Securities Act 1997

FORM 4

Act, Section 116(2)

NOTICE OF CHANGE IN SUBSTANTIAL SHAREHOLDING

Note: To be given to the company, the stock exchange on which it is listed and the Securities Commission by a person who is a substantial shareholder where there is a change of 1% or more in the relevant interests of the person. This form must be given within two days after the person becomes aware of the change. Use a separate form for each substantial shareholder that has a change of 1% or more in their relevant interests. Use a Form 5 where the person ceases to be a substantial shareholder as a result of the change.

1.	Name of listed company.	Oil Search Limited	
Note:	Insert the name of the listed company in which the substantial shareholding is held.		
2.	Name of stock exchange.	Port Moresby Stock Exchange Limited, Australia Stock Exchange	

Note: Insert the name of the stock exchange on which the company is listed.

3. Details of person that is a substantial shareholder.

Given names (natural persons only)	Surname or corporate name and registration number, if applicable	Residential address or address of registered office	Postal address
N/A	UBS AG	Bahnhofstrasse 45, 8001 Zurich Switzerland	Singapore Branch, One Raffles Quay, #50-01 North Tower, Singapore 048583

4. Date of change in relevant interests

11 July 2014	
--------------	--

Note: Insert the date of the change of the relevant interest of the person named in Item 3 in the listed company in Item 1, not the date on which the person became aware of the change.

5. Date of last notice given by substantial shareholder

12 March 2014

Note: Insert the date on which the person named it Item 3 last gave a notice in accordance with Section 115, 116 or 117 of the Act in relation to the voting shares i.e. a Form 3, 4 or 5.

6. Voting shares in which a relevant interest is held.

	Number of shares	% of shares in company.
Voting shares before the change	184,015,580	12.33%
Voting shares after the change	202,523,375	13.33 %

Note: Insert the number and percentage of voting shares in which the person has a relevant interest, before and after the date in Item 5. The terms "voting shares" and "relevant interest" are defined in Sections 112—114 of the Act.

7. Total number of voting shares issued by the listed company.

1,519,022,225	

Note: Insert the total number of voting shares issued by the listed company used for the purpose of calculating the percentages in Item 6.

8. Particulars of relevant interests after the change.

Class of voting shares in which relevant interest now held	Name(s) of registered holder(s)	Brief description of the nature of the relevant interests in the voting shares	Brief description of the transaction under which the relevant interests changed e.g. purchase / sale on stock exchange	Number of voting shares affected by the transaction	The consideration for the transaction (Kina)	Name of the other party to the transaction (if known) *	The consideration for the transaction
Ordinary	UBS AG	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	61,013 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Ordinary	UBS Fund Management (Switzerland) AG	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	206,070 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Ordinary	UBS Fund Services (Luxembourg) SA	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	84,948 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F

Ordinary	UBS Global Asset Management (Australia) Ltd	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	15,564,440 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Ordinary	UBS Global Asset Management Life Limited.	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	215,785 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Ordinary	UBS Global Asset Management (Singapore) Ltd	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	7,649 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Ordinary	UBS Global Asset Management (UK) Limited	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	2,021,102 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Ordinary	UBS AG Australia Branch	Prime Broker that has exercised its borrowing right in respect of shares pursuant to a Prime Broking Agreement (see Annexure E)	These holdings are held by UBS AG Australia Branch on behalf of Prime Brokerage clients	19,203,999 shares	Refer to Annexure F (Australian Dollars only)	Refer to Annexure E-1	Refer to Annexure F
Ordinary	UBS AG London Branch	Prime Broker that has exercised its borrowing right in respect of shares pursuant to a Prime Broking Agreement (see Annexure E)	These holdings are held by UBS AG London Branch on behalf of Prime Brokerage clients	16,250 shares	Refer to Annexure F (Australian Dollars only)	Refer to Annexure E-1	Refer to Annexure F
Ordinary	UBS Securities Australia Ltd	Beneficial Owner	Ordinary course transactions (e.g. on market purchases)	4,709,116 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Options	UBS Securities Australia Ltd	Beneficial Owner	Ordinary course transactions (e.g. on market purchases)	8,279,200 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Ordinary	UBS Wealth Management Australia Ltd	Broker with power to exercise discretion over account	Discretionary client accounts	2,763,559 shares	Refer to Annexure F (Australian Dollars only)	N/A	Refer to Annexure F
Ordinary	UBS Nominees Pty Ltd	See Section 11 below	See Section 11 below	137,012,250 shares	A\$1,123,500,450 (Australian Dollars only)	Independent State of Papua New Guinea	N/A
Ordinary	Independent State of Papua New Guinea	See Section 11 below	See Section 11 below	12,377,994 shares	A\$101,499,550.80 (Australian Dollars only)	Independent State of Papua New Guinea	N/A

^{*} This information is not required where the transaction took place on a stock exchange. Otherwise, if not known, insert "unknown".

Note: Where there is insufficient space on the form to supply the information required, use additional forms or annexe a separate sheet in the same format containing the information.

9. Details of other person(s) giving a notice for the same transaction.

Given names (natural persons only)	Surname or corporate name and registration number, if applicable	Residential address or address of registered office	Postal address
N/A			

Note: If known, insert the name and address of any other person(s) believed to have given, or to be intending to give, notice, whether a Form 3, 4 or 5, in relation to the transaction to which this notice relates. If not known, insert, "unknown".

10. Details of any associate with a relevant interest in the voting shares after the change.

Given names (natural persons only)	Surname or corporate name and registration number, if applicable	Residential address or address of registered office	Postal address
N/A	See item 8	Details of all UBS offices of following link: http://apps2	· · · · · · · · · · · · · · · · · · ·

Nature of Association:		
	i .	
Details of Relevant Interest of Associate:		

Note: Provide the name and address of each associate of the person submitting this notice who has a relevant interest in voting shares in the company, and the nature of the association with the associate and the details of the relevant interest of the associate.

11. Prescribed particulars of any agreement under which substantial shareholding acquired.

Date of any agreement	Material terms of any agreement (whether written or oral)
12 March 2014	The agreement under which UBS AG, Australia Branch (UBS) has acquired the relevant interests disclosed above in this Notice, as security for the finance arranged and provided by UBS, to the Independent State of Papua New Guinea (the State) under a bridge loan facility and an equity derivative financing, is constituted by the following documents:
	(a) Nominee Deed between the State, UBS, UBS Nominees Pty Ltd and UBS Securities Australia Pty Ltd (the Nominee Deed);
	(b) Specific Security Deed (CHESS Securities) between the State and UBS Nominees Pty Ltd;
	(c) Specific Security Deed (CHESS Securities - Collar) between the State and UBS Nominees Pty Ltd; and
	(d) 1995 ISDA Credit Support Annex, together with the agreed Elections and Variables in Paragraph 11 (the CSA).
	The material terms of the agreement giving rise to the relevant interests disclosed above in this Notice are summarised as follows.
	Under the Nominee Deed, UBS Nominees Pty Ltd acquired 137,012,250 shares in Oil Search Limited on trust for the State and to be transferred, dealt with or otherwise disposed of as the State may direct. UBS Securities Australia Limited is a party to this Deed as a market participant under the ASX Settlement Operating Rules.
	Separately, the State acquired in its own name 12,377,994 shares in Oil Search Limited.
	Under the two Specific Security Deeds, the State created security interests in the nature of fixed charges over the shares acquired in Oil Search Limited as follows:
	(i) under the Deed referred to at (b) above, a security interest over 12,377,994 in favour of UBS Nominees Pty Ltd (as Security Trustee for UBS as the initial lender under the bridge loan facility, the Arranger and the Facility Agent, and any other parties which may subsequently participate in the facility as lenders) to secure all amounts of moneys payable or owing under the bridge loan facility or any of the other transaction documents; and
	(ii) under the Deed referred to at (c) above, a security interest over any shares in Oil Search Limited held by UBS Nominees Pty Ltd as trustee for the State pursuant to the Nominee Deed, in favour of UBS to secure all amounts of money payable or owing under the equity derivative financing.
	Under the CSA, the title to the shares in Oil Search Limited which are the subject of the equity derivative financing and were initially held on trust for the State under the Nominee Deed were transferred to UBS as

eligible credit support for the financial obligations of the State under the equity derivative financing, and UBS has an obligation to transfer equivalent credit support (being shares in Oil Search Limited) back to the State upon settlement of the equity derivative financing.

UBS Nominees Pty Ltd and UBS Securities Australia Limited are both wholly owned subsidiaries of UBS and therefore bodies corporate which are related to UBS. As a result, any relevant interest in the shares of Oil Search Limited which either of those companies may have as a result of the agreement and documents summarised above is also a relevant interest for UBS in the shares of Oil Search Limited and have been treated as such in this Notice.

Other transactions undertaken pursuant to agreements entered into under market standard forms.

Note: Provide the details of the material terms of any agreement, whether written or oral and the details of the material terms of any relevant agreement through which a relevant interest, the subject of the notice, was changed. True copies of the documents must be submitted with this form.

12. The following documents must be submitted with this form

- A true copy of any document setting out the material terms of any agreement that:-A.
 - contributed to the person having to submit this form:
 - is in writing or in a document in which the material terms of an oral agreement have been reduced (b) to writing in accordance with Section 40(5)(c) of the Regulation; and
 - (c) is readily available to the person submitting the notice; or
- B. Where a document in Paragraph A is not readily available to the person submitting this form, a memorandum in writing detailing the material terms of the document. The number of pages that accompany, or are annexed to, this notice is N/A.

13. **Declaration**

I declare that to the best of my knowledge and belief the information in this form is true and correct.

Signature:

Full Name of Person Signing: Wuen-Yi-Toh

Role: Authorized Signatory

Date: 15 July 2014

Signature:

Full Name of Person Signing: Uma Tadela

Role: Authorized Signatory

Note: Where this notice is given by a body corporate it must be signed by a director or secretary who is authorised by the board of the body corporate to give the notice. Initials are not sufficient for the full name

of the person signing this form.



Nominee Deed

Independent State of Papua New Guinea

UBS AG, Australia Branch

ABN 47 088 129 613

UBS Nominees Pty Ltd

ABN 32 001 450 522

UBS Securities Australia Limited

ABN 62 008 586 481

Project Kumul

/2 March 2014

CONTENTS

CLAU	SE	P	AGE
1.	INTER	RPRETATION	3
	1.1 1.2	Definitions Interpretation - references	
2.	DECL	ARATION	5
	2.1 2.2 2.3 2.4 2.5 2.6 2.7	Declaration of Trust	6 6 6
3.	VOTIN	NG AT MEETINGS OF THE ISSUER	7
	3.1 3.2 3.3	Request Voting Transfer of Shares back to UBS	7
4.	UNDE	RTAKING	9
	4.1 4.2 4.3	Payment undertaking State undertakings Nominee undertakings	9
5.	POWE	RS AND DUTIES OF THE NOMINEE	9
	5.1 5.2 5.3 5.4	Limitations Transfer of Property Directions of the Court Knowledge of the Nominee	9 10
6.	OBLIG	ATIONS OF THE NOMINEE	10
	6.1 6.2 6.3 6.4 6.5	Extent of discretion	10 10 10
7.	LIMITA	ATION OF LIABILITY	. 11
	7.1 7.2 7.3 7.4 7.5 7.6 7.7	Indemnity Limitation of liability and obligations Nominee's lien as trustee Reliance on documents and experts Independent Investigation No monitoring Replacement of Nominee	.11 .11 .12 .12
8.	TAXES		. 13
	8.1 8.2 8.3 8.4 8.5	Obligation to pay Tax Notice to State Withholding Time for payment GST	. 13 . 13 . 14
9.	PAYME	NTS	. 14

10.	NOTICES		
	10.1 10.2 10.3 10.4 10.5	Notice by UBS and the Nominee	
11.	PROCESS AGENT16		
12.	PARTIC	IPANT	
13.	GENERAL		
	13.1 13.2 13.3 13.4 13.5 13.6	Stamp duty 18 Telephone recording 18 Amendments to deed 18 Waiver 18 Waiver of immunity 19 Governing Law 19	

THIS DEED is made on March 2014

BETWEEN

- (1) Independent State of Papua New Guinea (State)
- (2) **UBS AG, Australia Branch** ABN 47 088 129 613 (**UBS**)
- (3) **UBS Nominees Pty Ltd** ABN 32 001 450 522 (**Nominee**)
- (4) UBS Securities Australia Limited ABN 62 008 586 481 (Participant)

RECITALS

The Nominee agrees to act on the terms set out in this deed as trustee for the State.

THE PARTIES AGREE AS FOLLOWS:

1. INTERPRETATION

1.1 Definitions

Unless otherwise defined in this deed, or unless the context otherwise requires, capitalised terms and expressions used in this deed have the respective meanings given to them in the Confirmation (including by reference to one or more other documents). The following definitions also apply in this deed.

Amounts Payable means all amounts (including damages) that are payable, owing but not payable, or that otherwise remain unpaid by the State to UBS on any account at any time (whether present or future, actual or contingent) under or in connection with the Equity Derivative Financing Documents or the transactions which they contemplate.

ASIC means the Australian Securities and Investments Commission.

ASX Settlement means ASX Settlement Pty Ltd (ABN 49 008 504 532).

ASX Settlement Operating Rules means the operating rules of the clearing and settlement facility operated by ASX Settlement.

Beneficial Interest means the beneficial interest of the State in the Trust Property.

Confirmation means the long-form confirmation in respect of a step collar derivative transaction dated on or about the date of this deed between UBS and the State.

Corporations Act means the Corporations Act 2001 (Cth).

Encumbrance means:

- (a) a security interest that is subject to the PPS Act;
- (b) any other mortgage, pledge, lien or charge;
- (c) an easement, restrictive covenant, caveat or similar restriction over property; or
- (d) any other interest or arrangement of any kind that in substance secures the payment of money or the performance of an obligation, or that gives a creditor priority over unsecured creditors in relation to any property,

in each case, whether existing or agreed to be granted or created.

Government Agency means a government or government department, or governmental or semi-governmental person and a person (whether autonomous or not) charged with administration of any applicable law and includes without limitation the Australian Tax Office and any revenue authority of Papua New Guinea.

GST has the same meaning as given to that term in the GST Law.

GST Law has the meaning given to the term "GST law" in *A New Tax System (Goods & Services Tax) Act 1999* (Cth).

Indemnified Party means:

- (a) the Nominee;
- (b) UBS; and
- (c) any successor, substitute or assign of any of the above.

Nominee means the trustee of the Trust from time to time being UBS Nominees Pty Ltd (ABN 32 001 450 522) at the date of this deed.

Payment Undertaking has the meaning given to that term in clause 4.1 (*Payment undertaking*).

PPS Act means the Personal Property Securities Act 2009 (Cth).

Tax includes any tax, levy, impost, deduction, charge, rate, duty, compulsory loan or withholding which is levied or imposed by a Government Agency including GST, and any related interest, penalty, charge, fee or other amount, excluding stamp duty payable on this deed, or on the transfer of property to or by UBS or the Nominee.

Tax Invoice means a tax invoice complying with the requirements of any law about GST.

Trust means the trust constituted under this deed.

Trust Property means the sum of A\$10 and all other property transferred to and held by the Nominee pursuant to this deed from time to time (including any Shares) and all related rights, benefits and interests.

UBS Entity means UBS and any of its related bodies corporate.

1.2 Interpretation - references

Unless the contrary intention appears, a reference in this deed to:

- (a) (variations or replacements) a document (including this deed) includes any variation or replacement of it;
- (b) (clauses, annexures and schedules) a clause, annexure or schedule is a reference to a clause in or annexure or schedule to this deed;
- (c) (reference to statutes) a statute, ordinance, code or other law or the ASX Settlement Operating Rules includes regulations and other instruments under it and consolidations, amendments, re-enactments or replacements of any of them;
- (d) (law) law means common law, principles of equity, and laws made by parliament (and laws made by parliament include State, Territory and Commonwealth laws and regulations and other instruments under them, and consolidations, amendments, re-enactments or replacements of any of them);

- (e) (singular includes plural) the singular includes the plural and vice versa;
- (f) (person) the word "person" includes an individual, a firm, a body corporate, a partnership, a joint venture, an unincorporated body or association, or any government agency;
- (g) (executors, administrators, successors) a particular person includes a reference to the person's executors, administrators, successors, substitutes (including persons taking by novation) and assigns;
- (h) (two or more persons) an agreement, representation or warranty in favour of two or more persons is for the benefit of them jointly and each of them individually;
- (i) (jointly and severally) an agreement, representation or warranty by two or more persons binds them jointly and each of them individually except that the obligations of UBS and the Nominee under this deed are several and not joint;
- (j) (dollars) Australian dollars, dollars, A\$ or \$ is a reference to the lawful currency of Australia;
- (k) (meaning not limited) the words "including", "for example" or "such as" when introducing an example, do not limit the meaning of the words to which the example relates to that example or examples of a similar kind;
- (I) (time of day) a date or time is a reference to a date or time in Sydney;
- (m) (next Business Day) under this deed if an event must occur on a stipulated day which is not a Business Day then the stipulated day will be taken to be the next Business Day;
- (n) (headings) headings (including those in brackets at the beginning of paragraphs) are for convenience only and do not affect the interpretation of this deed; and
- (o) (**definitions**) a word or phrase defined in the Corporations Act or ASX Settlement Operating Rules, has the same meaning in this deed.

2. **DECLARATION**

2.1 **Declaration of Trust**

The Nominee declares that it holds each item of Trust Property as a trustee on the terms of this deed on trust for the State solely.

Subject to this deed, the Nominee undertakes to the State that it will:

- (a) transfer, deal with or otherwise dispose of the Trust Property as the State may direct and not otherwise; and
- (b) recognise the right of the State to dividends or distributions on Trust Property and will account to the State or as the State may direct for all Income or other distributions paid or made from time to time in respect of the Trust Property; and
- (c) deliver promptly or procure the transfer on demand to the State, or as it may direct, legal title to the Trust Property.

2.2 Present entitlement

The State is presently entitled to and has an absolute vested and indefeasible interest in the income of the Trust.

2.3 Termination

The Trust under this deed will terminate on the earlier of:

- (a) the date 80 years from the date of this deed; and
- (a) any earlier date determined by the Nominee in its absolute discretion with the consent of UBS.

2.4 No disposal

- (a) Except as expressly permitted by this deed, neither the Nominee nor the State shall dispose of any Trust Property during the continuation of the Trust.
- (b) In consideration for the transactions contemplated under the Equity Derivative Financing Documents, the State gives to UBS, and UBS has, irrevocable authority to give to the Nominee, on behalf of the State, such direction and instructions as to the disposal of Trust Property as the State is entitled to give under this deed. Without limiting the generality of this paragraph (b), this includes any such direction and instructions as to the disposal of Trust Property following UBS or any agent on its behalf appointing any controller or taking possession of any of the State's property (or any analogous event occurring under the law of any applicable jurisdiction).

2.5 **Beneficial interest**

- (a) The State is absolutely entitled pursuant to this deed as against the Nominee to the Trust Property.
- (b) The State must forebear from requiring the transfer of Trust Property to it or its nominee (other than the Nominee) prior to the termination date of the Trust.

2.6 Liability of Indemnified Parties

To the extent permitted by law, no Indemnified Party will be liable in respect of any conduct, delay, negligence or breach of duty in the exercise or non-exercise of any power, nor for any loss (including direct and indirect, economic and consequential loss) which results, except where it arises from fraud, gross negligence, or wilful misconduct by or on behalf of the Indemnified Party.

2.7 Encumbrances

- (a) The Nominee and the State must not create, or take or omit to take any action to permit to exist, any Encumbrance or other third party interest (other than an Encumbrance or third party interest created under an Equity Derivative Financing Document) over or in respect of any Trust Property or any interest in the Trust Property. Any such Encumbrance or other third party interest shall be invalid and unenforceable as between and/or against UBS, the Nominee and the State (or any person claiming under or through any of them).
- (b) Each of the Nominee and the State represents and warrants to UBS that no Trust Property is subject to any Encumbrance or other third party interest created by it (other than an Encumbrance or third party interest created under an Equity Derivative Financing Document).

3. VOTING AT MEETINGS OF THE ISSUER

3.1 Request

- (a) If the State is, or would be, eligible to vote at the meeting of members of the Issuer:
 - (i) at the record time for the members' vote; and
 - (ii) during the time when the Shares are proposed to be held by the Nominee in accordance with this clause 3 (*Voting at meetings of the Issuer*),

and provided that:

- (iii) no Event of Default or Termination Event has occurred in respect of the State; and
- (iv) an Ex-Distributions Date in respect of the Shares is not scheduled or anticipated to occur during the time when the Shares are proposed to be held by the Nominee in accordance with this clause 3 (Voting at meetings of the Issuer),

then the State may request in writing, at least 10 Business Days prior to the proposed transfer date, that UBS transfers to the Nominee a specified number of Shares which Shares must not exceed the number of Shares in the Collateral Balance of the State at the time of the request.

- (b) UBS will use its best commercial efforts (as determined by UBS) to comply with a written request from the State under paragraph (a), subject to the terms of the Equity Derivative Financing Documents.
- (c) The State must pay to UBS the stock borrow cost incurred by UBS to procure the delivery of Shares to be held by the Nominee, by no later than the third Business Day after the Shares are transferred to UBS as contemplated under clause 3.3 (*Transfer of Shares back to UBS*). The borrow cost shall be calculated on no more than the Delta Quantity of Shares.
- (d) To the fullest extent permitted by law, UBS has and is under no liability (whether in negligence or otherwise) to the State or the Nominee for any costs, liabilities, losses or expenses incurred or suffered by the State or the Nominee in connection with or relating to any ability or inability of the Nominee or the State to cast any vote at any meeting of members (or a class of members) of the Issuer if Shares are not delivered to the Nominee as contemplated by this clause 3 (Voting at meetings of the Issuer).

3.2 Voting

In respect of any Shares comprised in the Trust Property:

- (a) subject to paragraph (b), the Nominee shall not cast any vote at any meetings of members (or a class of members) of the Issuer; and
- (b) the State may direct the Nominee to cast a vote at any meeting of members (or a class of members) of the Issuer in respect of any such Shares, in which case the Nominee must vote in accordance with the direction of the State.

3.3 Transfer of Shares back to UBS

- (a) The State acknowledges and agrees that for so long as any Shares are held by the Nominee, the State has an outstanding obligation to re-deliver Equivalent Credit Support under the terms of the Confirmation.
- (b) Accordingly, the parties acknowledge and agree that the Shares are to be held by the Nominee for the shortest period of time necessary to give the State the entitlement to direct the Nominee in accordance with clause 3.2 (Voting) to vote at the meeting of members referred to in clause 3.1 (Request). The State hereby irrevocably and unconditionally instructs and directs the Nominee to transfer to UBS (which transfer is undertaken in the ordinary exercise of the Nominee's lien as trustee) all Shares held by the Nominee as soon as practicable and in any event within one Business Day after the record time for the member's vote.
- (c) Without limiting clause 3.1(a) or the other provisions of this clause 3.3 (Transfer of Shares back to UBS), if an Ex-Distributions Date for the Shares is scheduled or anticipated to occur at any time while the Shares are held by the Nominee, then the State hereby irrevocably and unconditionally instructs and directs the Nominee to transfer to UBS (which transfer is undertaken in the ordinary exercise of the Nominee's lien as trustee) all Shares held by the Nominee as soon as practicable and in any event prior to the relevant Ex-Distributions Date in respect of the Shares.
- (d) All Shares transferred to UBS in accordance with this clause 3.3 (*Transfer of Shares back to UBS*) will constitute the delivery of Eligible Credit Support under and for the purposes of the Confirmation.
- (e) The State must indemnify the Indemnified Parties against, and must pay to the Indemnified Parties on demand the amount of, all costs, liabilities, losses or expenses incurred by an Indemnified Party in respect of any delay in transfer of the Shares caused directly or indirectly by the State.
- (f) If, notwithstanding and without limiting paragraph (c):
 - (i) the Shares are held by the Nominee on an Ex-Distributions Date of a Distribution declared or resolved to be paid or made in respect of the Shares; and
 - (ii) the Nominee receives any Distribution,

then the State agrees that:

- (iii) the Nominee shall deliver to UBS, or pay to UBS an amount equal to, the entirety of such Distribution received from the Issuer by no later than the second Business Day after the making of such Distribution by the Issuer; and
- (iv) the provisions of the Confirmation in respect of Distributions in respect of Shares shall apply, as if the Shares remained held by UBS pursuant to the terms of the Confirmation, and with any other necessary changes.

The State hereby irrevocably and unconditionally instructs and directs the Nominee to make each such payment contemplated by this paragraph (f).

4. UNDERTAKING

4.1 Payment undertaking

At any time while the Trust Property consists of Shares, the Nominee indemnifies UBS against, and must pay to UBS on demand, the amount of all Amounts Payable (**Payment Undertaking**). The Payment Undertaking is a continuing indemnity and will (subject to the Nominee's limitation of liability in clause 7.2 (*Limitation of liability and obligations*)) extend to the ultimate balance of sums payable by the State under the Equity Derivative Financing Documents, regardless of any intermediate payment or discharge in whole or in part.

4..2 State undertakings

The State agrees:

- (a) to comply with all obligations affecting the Trust Property and pay on time all amounts for which the holder of the Trust Property is liable, including calls, instalments and Taxes;
- (b) at UBS's request in its discretion, to take up any rights attaching to Trust Property;
- (c) to promptly provide to UBS on request any information which UBS reasonably requests about the Trust Property or anything in relation to it; and
- (d) not to provide any instructions to the Nominee relating to the Trust Property that are inconsistent with this deed or any other Equity Derivative Financing Document.

4.3 Nominee undertakings

The Nominee and the State each acknowledges and agrees for the benefit of UBS in respect of each Share comprised in the Trust Property that:

- (a) the Nominee will not comply with instructions given by the State without seeking the consent of UBS except as expressly contemplated under clause 3 (Voting at meetings of members); and
- (b) the Nominee must comply with instructions (including instructions to transfer the Shares) given by UBS in accordance with the Equity Derivative Financing Documents without seeking the consent of the State.

5. POWERS AND DUTIES OF THE NOMINEE

5.1 **Limitations**

The Nominee shall have no powers, duties, rights or discretions in respect of the Trust except those expressly set out in this deed. In particular the Nominee shall have no power to dispose of Trust Property otherwise than in accordance with this deed.

5.2 Transfer of Property

Any power or duty of the Nominee to transfer any property under this deed (including the transfer of Trust Property) may be exercised or satisfied by instructing the Participant to transfer or dispose of the relevant property and/or posting or delivering to the registered address of the recipient an instrument of transfer of the relevant property in addition to any other mechanism or process permitted by law. Compliance by the Nominee with this clause will be a complete discharge of the Nominee's obligations with respect to the transfer of the relevant property.

5.3 **Directions of the Court**

UBS and/or the Nominee and/or the State may apply to a court for directions as to any matter arising in connection with the exercise of the powers and functions of UBS and/or the Nominee and/or the State under this deed, and the applicant shall not be responsible for any delay arising as a result.

5.4 Knowledge of the Nominee

The Nominee will only be considered to have knowledge, awareness or notice of a thing, or grounds to believe any thing, by virtue of the officers of the Nominee having day to day responsibility for the administration of the Trust having actual knowledge, actual awareness or actual notice of that thing, or grounds to believe that thing (and similar references will be interpreted in this way).

6. OBLIGATIONS OF THE NOMINEE

6.1 Extent of discretion

The Nominee need not:

- (a) act except when required to do so by this deed and then at the discretion of the Nominee as to the manner and time of acting; or
- (b) consult with the State before acting or giving any consent, approval or agreement or making any determination under this deed except where this deed expressly provides otherwise (including under clause 3.2 (*Voting*)).

6.2 In funds

The Nominee shall not act or fail to act if by doing so it will knowingly incur any material liability for which it is not satisfied that it is sufficiently indemnified from the Trust Property or otherwise.

6.3 No responsibility for deed

The Nominee has no responsibility for the form or contents of this deed or any other Equity Derivative Financing Document and will have no liability arising as a result of or in connection with any inadequacy, invalidity or unenforceability of any provision of this deed or any other Equity Derivative Financing Document.

6.4 Exclusion of further obligations

Despite any provision of this deed, the Nominee:

- (a) is not required to do or omit to do anything or incur any liability unless the Nominee's liability is limited in a manner satisfactory to the Nominee;
- (b) will not be under any obligations to advance or use its own funds for the payment of any costs, expenses or facilities;
- (c) will have no liability for the acts or omissions of any brokers or other agents whose acts or omissions are not reasonably capable of supervision by the Nominee;
- (d) has no responsibility for the performance by UBS and/or the State of their obligations under or in connection with this deed, the Trust Property, and the Nominee will have no liability arising as a result of or in connection with any act or omission of UBS and/or the State; and

(e) will have no obligations to prepare accounts or tax returns in respect of the Trust.

6.5 **Set Off**

The Nominee may set off or withhold any amount payable to it by UBS and/or the State against or from any amount payable by the Nominee to UBS and/or the State (as the case may be).

7. **LIMITATION OF LIABILITY**

7.1 Indemnity

- (a) The Nominee:
 - is fully indemnified out of the Trust Property in respect of any liability or obligation that the Nominee incurs while acting or purporting to act as Nominee; and
 - (ii) may reimburse itself from the Trust Property for all expenses incurred or payments made by it in respect of the Trust, except in relation to any liability, obligation or expense incurred as a result of the Nominee's fraud, gross negligence or wilful default.

The Trustee may apply either or both capital and income of the Trust Fund to satisfy its rights under this clause.

(b) Without limiting paragraph (a) above, the State acknowledges and agrees that the Nominee's liabilities and undertakings under this deed including in respect of the Payment Undertaking are expenses of the Trust for which the Nominee is entitled to be indemnified out of Trust Property. Accordingly, the State acknowledges and agrees that that if UBS makes a demand on the Nominee under the Payment Undertaking given under clause 4.1 (*Payment undertaking*), the Nominee (under its trustee's lien) may (without first notifying the State) dispose of the Trust Property and use the proceeds of that disposal to satisfy that demand.

7.2 Limitation of liability and obligations

- (a) Despite any other provision of this deed or any other Equity Derivative Financing Document, a liability or obligation of the Nominee arising under or in connection with this deed or any other Equity Derivative Financing Document (including in connection with the Payment Undertaking) is strictly limited to the extent to which (and can be enforced against the Nominee only to the extent to which) the Nominee is actually indemnified for the liability and the liability or obligation can lawfully be satisfied of the Trust Property. No person may seek to recover any shortfall in the amounts owing by the Nominee by bringing proceedings against the Nominee or applying to have the Nominee wound up.
- (b) When the Nominee, whether on instructions or under the Nominee's lien as trustee, exercise a power to dispose of Trust Property, the Nominee has no liability to the State for any delay in disposal or any failure to obtain a particular price or for obtaining difference prices for the Trust Property.

7.3 Nominee's lien as trustee

The State acknowledges and agrees that the Nominee's common law lien as trustee is a proprietary interest in the Trust Property that ranks in priority to the Beneficial Interest. However, nothing in this deed is intended to create a charge or security interest separate from the Trustee's rights at common law.

7.4 Reliance on documents and experts

The Nominee may rely on:

- any document (including any facsimile transmission or telegram) unless it knows or has reasonable grounds to believe that the document is not genuine and correct;
 and
- (b) advice, information and statements of UBS or other persons employed or retained by UBS,

and shall not be liable for any act or omission undertaken in reliance on that document, advice, information or statement. Without limitation, the Nominee may rely on such documents, advice, information and statements for the purpose of exercising and performing its powers and duties to make payments to the State and to transfer or dispose of Trust Property in accordance with this deed. Where the Nominee considers that it requires any advice, information or statement from any of the persons referred to in paragraph (b) of this clause, for the purpose of exercising or performing its powers and duties under this deed, the Nominee will have no liability in connection with any delay in the exercise or performance of the relevant powers or duties pending receipt of the relevant advice, information or statements.

7.5 **Independent Investigation**

The State confirms that it has made and will continue to make, independently and without reliance on the Nominee or UBS:

- (a) its own investigations into the affairs of UBS and the Nominee; and
- (b) its own analyses and decisions whether to take or not take action under this deed.

7.6 **No monitoring**

The Nominee is not required to keep itself informed as to any Issuer or the compliance by UBS and/or the State with this deed or any other document or agreement or to inspect any property or book of UBS and/or the State.

7.7 Replacement of Nominee

- (a) Subject to the appointment of a successor Nominee as provided in this clause:
 - (i) the Nominee may resign at any time by giving not less than 30 days written notice to UBS and/or the State; and
 - (ii) UBS may remove the Nominee from office if the Nominee breaches any of its obligations under this deed and the breach remains unremedied after UBS has given not less than 30 days written notice to the Nominee.
- (b) On notice of resignation or removal UBS has the right to appoint a successor Nominee who accepts the appointment.
- (c) If no successor Nominee is appointed within 30 days after notice, the retiring Nominee may appoint a successor Nominee who accepts the appointment.
- (d) On its appointment the successor Nominee will have all the rights, powers and obligations of the retiring Nominee (except liabilities arising from defaults).
- (e) The retiring Nominee shall execute and deliver all documents or agreements which UBS considers necessary or desirable in UBS's opinion to transfer to the successor

Nominee this deed or to effect the appointment of the successor Nominee (subject to satisfaction of all liabilities owed to the Nominee on its own account under this deed or for which the Nominee may be personally liable).

- (f) After any retiring Nominee's resignation or removal, this clause will continue in effect in respect of anything done or omitted to be done by it while it was acting as Nominee, and the retiring Nominee retains the rights and remedies available to it under this deed or at law in relation to the performance and exercise of its powers, duties and functions while Nominee.
- (g) UBS shall not unreasonably withhold its approval of any proposed successor Nominee. It shall respond as soon as practicable to any request for approval.
- (h) The State agrees not to remove the Nominee as its nominee under this deed during the term of the Equity Derivatives Financing Documents.

8. TAXES

8.1 **Obligation to pay Tax**

- (a) Where the Nominee has received notice requiring the payment of any Tax on behalf or in respect of the State or Trust Property, the Nominee shall determine, on advice, whether it has any obligation to pay or provide for the amount of the Tax. The Nominee may also at any time obtain advice as to whether it has any obligation to pay any Tax on behalf of or in respect of the State or any Trust Property, whether or not it has received notice requiring payment. If the Nominee determines that it has an obligation to pay the Tax, or there are reasonable grounds to argue that it has an obligation to pay the Tax, the Nominee shall give notice to the State, with a copy of that notice to be provided to UBS.
- (b) The State agrees to indemnify the Nominee in respect of any tax referred to in clause 8.1(a).

8.2 Notice to State

A notice given by the Nominee to the State pursuant to clause 8.1 (*Obligation to pay Tax*) shall set forth reasonable particulars pertaining to the obligation to make the relevant Tax payment.

8.3 Withholding

- (a) If the Nominee receives advice that it is prudent to withhold any amount of Tax in respect of any amount otherwise payable to the State then, notwithstanding any other provision in this deed, the Nominee will withhold that amount for so long as it remains prudent to do so.
- (b) Despite any other provision in this deed, if at any time an applicable law obliges the Nominee to make a deduction or withholding in respect of Taxes from a payment under this document, the Nominee:
 - (i) must notify the State of the obligation (with a copy of that notification to be provided to UBS);
 - (ii) must ensure that the deduction or withholding does not exceed the minimum amount required by law; and
 - (iii) must pay to the relevant Government Agency on time the full amount of the deduction or withholding and promptly deliver to the State a copy of any receipt, certificate or other proof of payment.

(c) If any amount is deducted or withheld in accordance with paragraph (b) above, the Nominee is not required to gross up any payment or future payment either under this deed or any other Equity Derivative Document to the State in respect of that amount deducted or withheld.

8.4 Time for payment

The State must make any payment due pursuant to the indemnity in clause 8.1(b) on or before the day that is five Business Days before the last date on which payment of the relevant Tax may lawfully be made without incurring penalties or interest for late payment.

8.5 **GST**

If GST is or will be payable on a supply by UBS or the Nominee made under or in connection with this deed, to the extent that the consideration otherwise provided for that supply under this deed is not stated to include an amount in respect of GST on the supply, the State must:

- (a) pay to UBS or the Nominee an additional amount equal to any GST which UBS or the Nominee must pay for any supply by UBS or the Nominee to the State under this deed:
- (b) pay that additional amount as and when the State must pay the other consideration for the supply; and
- (c) indemnify UBS and the Nominee from any GST and against any damage or cost directly or indirectly arising from or caused by the failure of the State to pay any amount as and when required by this clause.

UBS or the Nominee will issue the State with a Tax Invoice in respect of any additional amount on account of GST that UBS or Nominee may recover from the State under this clause.

9. **PAYMENTS**

All monies payable under this deed shall be paid by electronic funds transfer to an account designated by the payee unless otherwise agreed.

10. **NOTICES**

10.1 Notice by UBS and the Nominee

Except where otherwise provided in this deed, all notices required or permitted to be given by UBS or the Nominee to the State pursuant to this deed must be in writing and will be duly given if:

- (a) left at the State's address as set out in clause 10.5 (Addresses for notices); or
- (b) sent by prepaid mail to the State's address (which must be airmail if that address is not within Australia);
- (c) sent by facsimile transmission to the State's facsimile number; or
- (d) sent in electronic form (such as email), with the subject matter line of an email specifying the State's name, this deed and a brief description of the subject matter of the communication.

10.2 Time of receipt

A notice given by UBS or the Nominee in accordance with clause 10.1 is treated as having been duly given and received:

- (a) when delivered, in the case of a notice left at the State's address;
- (b) on the second Business Day after posting, in the case of being sent by pre-paid mail to an address in Australia;
- (c) on the fifth Business Day after posting, in the case of being sent by pre-paid mail to an address outside Australia; and
- (d) at the time of confirmation of transmission, in the case of being sent by facsimile transmission; and
- (e) when sent in readable form:
 - (i) by 5.00 pm (local time in the place of receipt) on a Business Day on that Business Day; or
 - (ii) after 5.00 pm (local time in the place of receipt) on a Business Day, or on a day that is not a Business Day – on the next Business Day,

in the case of being sent in electronic form.

10.3 Notice by State

All notices required or permitted to be given by the State to UBS or the Nominee pursuant to this deed must be in writing and will be treated as being duly given if and only if they are actually received by UBS or the Nominee at the address of that party as set out in clause 10.5 (*Addresses for notices*) or at such other address as may have been notified to the State.

10.4 Notices Between UBS and Nominee

Notices required or permitted to be given by UBS to the Nominee or by the Nominee to UBS may be given orally provided UBS and the Nominee are related bodies corporate.

10.5 Addresses for notices

State

Address: The Independent State of Papua New Guinea

Department of Treasury

P O Box 542 Waigani 131 N.C.D.

Papua New Guinea

Telephone: +675 3235600

Fax number: +675 3128804

Email: dairi_vele@treasury.gov.pg

Attention: Mr Dairi Vele, Secretary for Department of Treasury

UBS

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61-2-9324-2880

Email: <u>elaine.chow@ubs.com</u> / <u>kevin.bayett@ubs.com</u> / <u>sh-aus-eq-</u>

support@ubs.com / michael.rollin@ubs.com / sh-syd-rmp-op@ubs.com

Attention: Elaine Chow / Kevin Bayett / Michael Rollin

Nominee

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61-2-9324-2880

Email: <u>elaine.chow@ubs.com</u> / <u>kevin.bayett@ubs.com</u> / <u>sh-aus-eq-</u>

support@ubs.com / michael.rollin@ubs.com / sh-syd-rmp-op@ubs.com

Attention: Elaine Chow / Kevin Bayett / Michael Rollin

Participant

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61-2-9324-2880

Email: <u>elaine.chow@ubs.com</u> / <u>kevin.bayett@ubs.com</u> / <u>sh-aus-eq-</u>

 $\underline{support@ubs.com} \ / \ \underline{michael.rollin@ubs.com} \ / \ \underline{sh-syd-rmp-op@ubs.com}$

Attention: Elaine Chow / Kevin Bayett / Michael Rollin

11. PROCESS AGENT

- (a) The State irrevocably appoints Norose Notices Australia Pty Ltd ACN 158 029 586 of Level 16, Grosvenor Place, 225 George Street, Sydney, Australia as its agent under this deed for service of process in any proceedings in the State of New South Wales.
- (b) If any person appointed as process agent is unable for any reason to act as agent for service of process, the State must immediately appoint another agent on terms acceptable to UBS and the Nominee. Failing this, UBS may appoint another person for this purpose.

- (c) The State acknowledges that service upon the process agent appointed under this clause shall not constitute a breach of the Vienna Convention on Diplomatic Relations and/or the Vienna Convention on Consular Relations.
- (d) The State undertakes to maintain the appointment of the process agent until all money owed by the State to UBS is paid to UBS, and agrees that any process served on that person is taken to be served on it.
- (e) This clause does not affect any other method of service allowed by law.

12. PARTICIPANT

- (a) Each of the State and the Nominee notify the Participant that the State has granted or will grant a security interest in its present and future right, title and interest in, to, under and derived from all Shares transferred to, and held by, the Nominee as trustee for the State pursuant to this deed (**Security**).
- (b) Each of the State and the Nominee irrevocably and unconditionally:
 - (i) authorise any representative of UBS (the **Authorised Signatories**) severally to be the sole signatories authorised on behalf of the State and the Nominee to instruct the Participant from time to time in relation to the Shares (except as contemplated by clause 3.2);
 - (ii) (except as contemplated by clause 3.2) direct the Participant to only act on the instructions and directions of an Authorised Signatory in respect of the Shares, including instructions and directions by which the Shares can be transferred or otherwise dealt with, and to only act in accordance with any such instructions and/or directions from the Authorised Signatories including the provision of any information that they request on the Participant's holding of the Shares (for the avoidance of doubt, any such instructions or directions may include the giving of communications by means of electronic messages or other electronic communications); and
 - (iii) direct the Participant to pay or deliver (as appropriate):
 - (A) the proceeds of enforcement of the Security; or
 - (B) (subject to any contrary directions of UBS) any dividends, rights, payments, distributions, other amounts and other entitlements to which UBS is entitled under the terms of any Equity Derivative Financing Document in respect of any Shares,

into the following account (Nominated Bank Account):

Bank: To be advised by UBS to the other parties
Account name: To be advised by UBS to the other parties
SWIFT ID: To be advised by UBS to the other parties
Account no: To be advised by UBS to the other parties
Payment directions: To be advised by UBS to the other parties,

or as an Authorised Signatory may direct from time to time.

- (c) The Participant confirms that it:
 - (i) accepts the terms, authorisations and directions contained in paragraph (b) and undertakes to act only in accordance with the instructions and directions of an Authorised Signatory (including where such instructions relate to the sending of electronic messages or other electronic communications) in

relation to the Shares and will provide information on the Shares to an Authorised Signatory upon request;

- (ii) will pay any proceeds of:
 - (A) enforcement of the Security; or
 - (B) (subject to any contrary directions of UBS) any dividends, rights, payments, distributions and other amounts to which UBS is entitled under the terms of any Equity Derivative Financing Document in respect of any Shares,

into the Nominated Account or as an Authorised Signatory may direct from time to time.

(d) Each of the State, the Nominee and UBS acknowledge that the Participant is entitled to assume the genuineness and authenticity of any instruction purported to be given by an Authorised Signatory and each of the State and the Nominee is deemed to have ratified and confirmed any such instruction. The Participant is not liable for any act or loss that results from the Participant having acted on an instruction given by a person it reasonably believes to be an Authorised Signatory for the purposes of this clause.

13. GENERAL

13.1 Stamp duty

The State shall be responsible for the payment of all stamp duties and documentary taxes (including any interest or penalty that is payable) assessed by any revenue authority in Australia and/or Papua New Guinea upon the transfer of Trust Property to or by the Nominee, the constitution of the Trust, the execution of this deed or any other transactions contemplated by this deed.

13.2 Telephone recording

Each State agrees to:

- (a) the tape recording by UBS or its agents of any telephone conversation with the State;
- (b) the retention of any tape recording so made; and
- (c) the use of any tape recording so made as evidence of the contents of the telephone conversation.

13.3 Amendments to deed

UBS, the State and the Nominee may amend this deed from time to time by supplemental deed signed by the parties.

13.4 Waiver

No failure to exercise and no delay in exercising any power operates as a waiver. Nor does any single or partial exercise of any power preclude any other or further exercise of that power or any other power. The powers in this deed are in addition to and do not exclude or limit any right, power or remedy provided by law.

13.5 Waiver of immunity

Each party irrevocably waives, to the extent permitted by applicable law, with respect to itself and its revenues and assets (irrespective of their use or intended use), all immunity on the grounds of sovereignty or other similar grounds from:

- (a) suit;
- (b) jurisdiction of any court;
- (c) relief by way of injunction or order for specific performance or recovery of property;
- (d) attachment of its assets (whether before or after judgment); and
- (e) execution or enforcement of any judgment,

to which it or its revenues or assets might otherwise be entitled in any Proceedings in the courts of any jurisdiction and irrevocably agrees, to the extent permitted by applicable law, that it will not claim any such immunity in any Proceedings.

13.6 **Governing Law**

- (a) This deed shall be governed by the laws applying in New South Wales.
- (b) The parties submit to the exclusive jurisdiction of the courts of New South Wales.
- (c) Nothing in this clause 13.6 limits the right of UBS or the Nominee to institute proceedings arising out of or in connection with this deed or any Beneficial Interest against the State in any manner permitted by law.

EXECUTED as a deed.

Signed, Sealed and Delivered for and on behalf of the INDEPENDENT STATE OF PAPUA NEW GUINEA by Sir Michael Ogio, G.C.M.G., K.St.J., Governor General, acting with and in accordance with the advice of the National Executive Council:



in the presence of:

		2c
Navasi		
wame:	CARL	عاد مرا <i>ح</i> د
Title:	I ALL LIER	

SIGNED, SEALED and **DELIVERED** for **UBS AG, AUSTRALIA BRANCH** by its duly authorised signatories:

Signature	Signature
Name:	Name:
Position:	Position:
SIGNED, SEALED and DELIVERED for UBS NOMINEES PTY LIMITED by its duly appointed attorneys:	
Signature	Signature
Name of attorney:	Name of attorney:

EXECUTED as a deed.

Signed, Sealed and Delivered for and on behalf of the **INDEPENDENT STATE OF PAPUA NEW GUINEA** by Sir Michael Ogio, G.C.M.G., K.St.J., Governor General, acting with and in accordance with the advice of the National Executive Council:

with and in accordance with the advice of the National Executive Council: in the presence of: Name: Title: SIGNED, SEALED and DELIVERED for UBS AG, AUSTRALIA BRANCH by its duly authorised signatories: Signature Name: Name: Position: Position: SIGNED, SEALED and DELIVERED for UBS NOMINEES PTY LIMITED by its duly appointed attorneys: Signature Name of attorney:

Signature

Kevin Bayett B140 Larror

Name of attorney:

KEVIN LU

SIGNED, SEALED and DELIVERED for UBS SECURITIES AUSTRALIA LIMITED

by its duly appointed attorneys:

Signature

Signature

Ken is Bayet

BRAD GORGON

Name of attorney:

Name of attorney:

Name :

KEVIN LU

Specific Security Deed (CHESS Securities)

Independent State of Papua New Guinea

and

UBS Nominees Pty Ltd

ABN 32 001 450 522

Project Kumul

12 MARCH 2014

CONTENTS

CLAUSE			
1.	INTERPRETATION		
	1.1 1.2 1.3 1.4 1.5	Definitions	
2.	SECUR	VITY5	
	2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8 2.9	Security clause	
3.	DEALI	NGS WITH SECURED PROPERTY7	
	3.1	Restricted dealings7	
4.	REPRE	SENTATIONS AND WARRANTIES7	
	4.1 4.2 4.3 4.4	General representations and warranties of the Grantor	
5.	GRANT	OR'S UNDERTAKINGS8	
	5.1 5.2 5.3	Undertakings of the Borrower	
6.	SECUR	ED PARTY'S POWERS10	
	6.1 6.2 6.3 6.4	Exercise of rights by Secured Party	
7	POWER	R OF ATTORNEY11	
	7.1 7.2	Appointment of Attorneys	
8.	ENFOR	CEMENT	
	8.1 8.2 8.3 8.4	Circumstances when this document may be enforced	
9.	DEFAU	LT INTEREST	
	9.1 9.2 9.3	Grantor must pay interest	

10.	APPOINTMENT OF RECEIVER			
	10.1 10.2 10.3	Appointment		
11.	PROTECTION OF SECURED PARTY AND APPOINTEES			
	11.1 11.2 11.3	Protection of Secured Party and Receiver		
12.	PROTECTION OF THIRD PARTIES			
	12.1 12.2	Dealings under this document		
13.	APPLICATION OF MONEY			
	13.1 13.2 13.3 13.4 13.5 13.6 13.7	Order18Only actual receipts credited18Compensation18Certificates and disputes19No interest19Payment into bank account19Amounts contingently due19		
14.	CONTI	CONTINUING SECURITY		
	14.1 14.2 14.3 14.4	Continuity19Limitations on Grantor's rights20No marshalling20Effect of Insolvency Event20		
15.	RECOV	ERY OF GST		
16.	6. INDEMNITIES			
	16.1 16.2 16.3 16.4	General indemnities21Indemnity for exercise of rights or proceedings22Currency indemnity22Recovery from Secured Property22		
17.	DISCH	ARGE23		
18.	CONFI	DENTIALITY		
	18.1 18.2	General confidentiality		
19.	NOTICES24			
	19.1 19.2 19.3 19.4	How to give a notice		
20.	AMENDMENT AND ASSIGNMENT25			
	20.1 20.2	Amendment 25 Assignment 25		
21.	GENERAL			
	21.1 21.2 21.3 21.4	Governing law		

	21.5	Variation of rights	27
	21.6	Operation of this document	27
	21.7	Operation of indemnities	27
	21.8	Consents	28
	21.9	Statements by Secured Party	28
	21.10	Set-off	28
	21.11	No merger	28
	21.12	Exclusion of contrary legislation	28
	21.13	Counterparts	28
	21.14	Execution by fewer than all parties	28
Sched			
1	Initial S	Securities	29

BETWEEN:

- (1) Independent State of Papua New Guinea (the Grantor); and
- UBS Nominees Pty Ltd ABN 32 001 450 522 of Level 16, Chifley Tower, 2 Chifley Place, (2) Sydney NSW 2000, Australia (the Secured Party).

RECITALS

- (A) The parties entered into the Facility Agreement on or around the date of this document.
- (B) The parties enter into this document in connection with the Facility Agreement.

THE PARTIES AGREE AS FOLLOWS:

1. INTERPRETATION

1.1 Definitions

The following definitions apply in this document.

ASX Settlement means ASX Settlement Pty Limited (ABN 49 008 504 532).

ASX Settlement Rules means the ASX Settlement Operating Rules and any other operating rules, procedures, directions, decisions, requirements, customs, usages and practices of ASX Settlement, as amended from time to time.

Attorney means an attorney appointed under a Transaction Document and any attorney's substitute or delegate.

CHESS Security means a Marketable Security that is registered on a CHESS Subregister.

CHESS Subregister means, for the Issuer, that part of its share register that is maintained by ASX Settlement under CHESS.

Collateral Security means a Security Interest or Guarantee (other than the Security Interest granted in clause 2.1) from any person that secures or otherwise provides for payment of any Secured Money.

Controlling Participant means UBS Securities Australia Limited ABN 62 008 586 481 or any other person acceptable to the Secured Party who is able to sponsor Marketable Securities on a CHESS Subregister.

Event of Default means an "Event of Default" as defined in the Facility Agreement.

Facility Agreement means the Bridge Facility Agreement of that title dated on or about the date of this document and made between the Grantor, UBS AG, Australia Branch and the Secured Party.

Initial Security means a Marketable Security of the Issuer specified in Schedule 1.

Issuer means Oil Search Limited, a company incorporated under the laws of the Independent State of Papua New Guinea with ARBN 055 079 868.

Marketable Security means:

an intermediated security; (a)

- (b) a debenture, stock or bond, unit in a unit trust, or other marketable security; and
- (c) any other investment instrument,

New Right means a present or future right of the Grantor:

- to or in any money, dividend (including any return of capital), interest, offer, bonus, note or other Marketable Security, or any entitlement to subscribe for any of them;
- resulting from any substitution, conversion, redemption, forfeiture, cancellation, reclassification, consolidation or subdivision; or
- (c) resulting from a reduction of capital, liquidation or scheme of arrangement,

in connection with the Secured Property.

Receiver means a receiver or a receiver and manager.

Secured Money means all amounts (including damages) that are payable, owing but not payable, or that otherwise remain unpaid by the Grantor to the Secured Party and the Finance Parties on any account at any time under or in connection with the Facility Agreement or any other Transaction Document:

- (a) whether present or future, actual or contingent;
- (b) whether incurred alone, jointly, severally or jointly and severally;
- (c) whether the Grantor is liable on its own account or for the account of, or as surety for, another person and without regard to the capacity in which the Grantor is liable;
- (d) whether due to the Secured Party alone or with another person;
- (e) whether the Secured Party is entitled for its own account or the account of another person;
- (f) whether arising from a banker and customer relationship or any other relationship;
- (g) whether originally contemplated by the Grantor or the Secured Party or not;
- (h) whether the Secured Party is the original person in whose favour the undertakings in this document, the Facility Agreement or any other Transaction Document were given or an assignee and, if the Secured Party is an assignee:
 - (i) whether or not the Grantor consented to or knew of the assignment;
 - (ii) no matter when the assignment occurred; and
 - (iii) whether or not the entitlements of that original person were assigned with the Security Interest granted in clause 2.1; and
- (i) if determined pursuant to any award, order or judgment against the Grantor, whether or not the Grantor was a party to the court proceedings, arbitration or other dispute resolution process in which that award, order or judgment was made.

Secured Property means all the Grantor's present and future interest in, to, under or derived from all:

- (a) Initial Securities; and
- (b) New Rights.

Sponsorship Agreement means the Participant Sponsorship Agreement dated on or about the date of this document between the Grantor, the Secured Party and the Controlling Participant, under which the Controlling Participant is the sponsor of the CHESS Security, and which complies with the ASX Settlement Rules.

Transfer means, in respect of a Marketable Security or a New Right, an executed document of transfer sufficient to transfer all the legal and beneficial ownership of that Marketable Security or New Right to the Secured Party or its nominee.

1.2 Definitions in ASX Settlement Rules

Unless otherwise defined in this document, any capitalised term used in this document which is defined in the:

- (a) ASX Settlement Rules has the meaning given in the ASX Settlement Rules; and
- (b) Security Trust Deed, including by incorporation, has the same meaning given in the Security Trust Deed.

1.3 Rules for interpreting this document

Headings are for convenience only, and do not affect interpretation. The following rules also apply in interpreting this document, except where the context makes it clear that a rule is not intended to apply.

(a) A reference to:

- (i) a legislative provision or legislation (including subordinate legislation) is to that provision or legislation as amended, re enacted or replaced, and includes any subordinate legislation issued under it;
- (ii) a document (including this document) or agreement, or a provision of a document (including this document) or agreement, is to that document, agreement or provision as amended, supplemented, replaced or novated;
- (iii) a party to this document or to any other document or agreement includes a successor in title, permitted substitute or a permitted assign of that party;
- (iv) a person includes any type of entity or body of persons, whether or not it is incorporated or has a separate legal identity, and any executor, administrator or successor in law of the person; and
- (v) anything (including a right, obligation or concept) includes each part of it and any part of it.
- (b) A singular word includes the plural, and vice versa.
- (c) A word which suggests one gender includes the other genders.
- (d) If a word or phrase is defined, any other grammatical form of that word or phrase has a corresponding meaning.
- (e) If an example is given of anything (including a right, obligation or concept), such as by saying it includes something else, the example does not limit the scope of that thing.

- (f) The word **agreement** includes an undertaking or other binding arrangement or understanding, whether or not in writing.
- (g) The expression **this document** includes the agreement, arrangement, understanding or transaction recorded in this document.
- (h) An Event of Default "continues" until the Secured Party notifies the Grantor that it has been:
 - (i) remedied to the satisfaction of the Secured Party; or
 - (ii) waived by the Secured Party.
- (i) Unless otherwise defined in this document, words that are defined in an applicable Accounting Standard have the same meaning in this document.
- (j) Unless otherwise defined in this document, words defined in the GST Law have the same meaning in clauses concerning GST.
- (k) If a person is a member of a GST group, references to GST for which the person is liable and to input tax credits to which the person is entitled include GST which the representative member of the GST group is liable and input tax credits to which the representative member is entitled.

1.4 PPSA terms

In this document, unless the context requires otherwise, the following terms have the meanings given to them in the PPSA:

- (a) amendment demand;
- (b) attach;
- (c) control;
- (d) financing change statement;
- (e) financing statement;
- (f) intermediated security;
- (g) investment instrument;
- (h) registration; and
- (i) verification statement.

1.5 Non Business Days

If the day on or by which a person must do something under this document is not a Business Day:

- (a) if the act involves a payment that is due on demand, the person must do it on or by the next Business Day; and
- (b) in any other case, the person must do it on or by the previous Business Day.

1.6 **Security Trustee provisions**

Clause 4 (*Limitation on recourse to Security Trustee*) of the Security Trust Deed applies to this document as if set out in full (but as if all references in that clause to "this document" were references to this document).

2. **SECURITY**

2.1 Security clause

The Grantor grants a security interest in the Secured Property to the Secured Party to secure the payment of the Secured Money. This security interest is a fixed charge.

2.2 **Priority**

The Security Interest granted in clause 2.1 is intended to take effect as a first ranking security subject only to those Permitted Security Interests which the Secured Party agrees in writing rank in priority to it.

2.3 Initial Securities

The Grantor must at the time it executes this document:

- (a) enter into the Sponsorship Agreement with the Secured Party and the Controlling Participant in respect of the Initial Securities and terminate all other sponsorships of the Initial Securities;
- (b) give the Secured Party and the Controlling Participant the applicable holder or other identification numbers and information sufficient to identify the Initial Securities (including a copy of the relevant holding statement); and

(c)

- (i) do everything within its power that is necessary to ensure that the Initial Securities are recorded on the Issuer's CHESS Subregister; and
- (ii) do everything necessary to transfer sponsorship of the Initial Securities to the Controlling Participant.

This clause does not apply to the extent that the Secured Party agrees otherwise under clause 2.9.

2.4 Documents relating to New Rights

The Grantor must deposit with the Secured Party or the Secured Party's nominee:

- (a) all certificates, transfers and other documents or agreements evidencing title to New Rights; and
- (b) Transfers (in form and number satisfactory to the Secured Party) with the name of the transferee, the consideration and the date left blank, in respect of those new Rights,

immediately it acquires those New Rights.

2.5 **Dividends and votes**

The Grantor may:

- (a) exercise all rights (including voting rights) in connection with the Secured Property;
- (b) (subject to this document) receive and retain free of the Security Interest granted in clause 2.1 the benefits of all cash dividends (other than returns of capital) and other income comprised in New Rights which are distributed to holders of the Marketable Securities,

except while an Event of Default continues.

2.6 Secured Party entitled to exercise rights on default

If an Event of Default occurs and is continuing:

- (a) the Grantor must, if the Secured Party gives notice to the Grantor requiring it to do so, do everything necessary to ensure that the Secured Property is registered in the name of the Secured Party or its nominee in accordance with any directions contained in that notice:
- (b) the Grantor's rights under clause 2.5 immediately cease;
- (c) the Secured Party may exercise or refrain from exercising any rights in connection with the Secured Property; and
- (d) the Secured Party may receive all New Rights and apply them (or their sale proceeds) in accordance with clause 13.1.

2.7 Acknowledgment of no subordination

The Grantor acknowledges that the Secured Party has not agreed to subordinate its Security Interest in the Secured Property to any other interest in the Secured Property, except to the extent (if any) expressly provided by a Transaction Document.

2.8 Collateral Security

The Security Interest granted in clause 2.1 is collateral to and secures the same money as is secured by Collateral Securities.

2.9 Nominees and custodians

The Grantor must hold all Secured Property in its own name, and not through a nominee or custodian. If the Secured Party agrees that any of the Secured Property can be held by a nominee or custodian on behalf of the Grantor, the Grantor undertakes to procure that the nominee or custodian:

- (a) acknowledges and consents to the Security Interest created by clause 2.1;
- (b) agrees to hold the Secured Property on such terms as the Secured Party requires, including by agreeing not to:
 - create or allow another interest in any Secured Property other than a Permitted Security Interest; or
 - (ii) dispose, or part with possession, of any Secured Property,

without the consent of the Secured Party; and

(c) agrees to do anything, including executing documents, that the Secured Party may require to enable it to perfect its Security Interest by control.

3. **DEALINGS WITH SECURED PROPERTY**

3.1 Restricted dealings

The Grantor must not do, or agree to do, any of the following unless it is permitted to do so by another provision in a Transaction Document:

- (a) create or allow another interest in any Secured Property other than a Permitted Security Interest; or
- (b) dispose, or part with possession, of any Secured Property.

4. REPRESENTATIONS AND WARRANTIES

4.1 General representations and warranties of the Grantor

Clause 12 of the Facility Agreement is incorporated into this document as if set out in full, except that references to the "Borrower" are to the "Grantor".

4.2 Representations and warranties regarding Secured Property

The Grantor represents and warrants to the Secured Party that:

(a) (Secured Property)

- (i) it is the legal and beneficial owner of the Secured Property (except to the extent that any of the Secured Property is held by a nominee or custodian in accordance with clause 2.9), and
- (ii) it has full power to grant a Security Interest in the Secured Property in the manner provided in this document; and
- (iii) the Security Interest granted in clause 2.1 is an effective security in the Secured Property, except to the extent (if any) set out in any Transaction Document or that the Secured Party may otherwise agree;
- (b) (**no other interest**) no other person has any interest in or other right over the Secured Property except:
 - (i) to the extent (if any) set out in any Transaction Document;
 - (ii) Permitted Security Interests; or
 - (iii) as otherwise agreed by the Secured Party;
- (c) (**location of Secured Property**) at the date of execution of this document, none of the Secured Property is located, or taken for the purposes of New South Wales stamp duty law to be located, in New South Wales;
- (d) (no escrow or other conditions) the Secured Property is not subject to any escrow or other conditions imposed by the Corporations Act or under the rules of any stock exchange;
- (e) (all calls satisfied) all calls made in respect of the Secured Property have been satisfied;
- (f) (fully paid Marketable Securities) the Marketable Securities in the Secured Property are all fully paid; and

(g) (**no money owing to Issuer**) no money is owing by the Grantor to the Issuer in connection with the Secured Property.

4.3 Repetition of representations and warranties

The representations and warranties in this clause are taken to be repeated on each date from the date of this document until the Termination Date (as defined in the Facility Agreement) or any date after the Termination Date (as defined in the Facility Agreement) on which all amounts outstanding under the Transaction Documents have been paid in full, on the basis of the facts and circumstances as at that date.

4.4 No representations by the Secured Party

The Grantor acknowledges that it has not relied and will not rely on any financial or other advice, representation, statement or promise provided or made by or on behalf of the Secured Party in deciding to enter into this document or to exercise any right or perform any obligation under it.

5. GRANTOR'S UNDERTAKINGS

5.1 Undertakings of the Borrower

Clause 13 of the Facility Agreement is incorporated into this document as if set out in full, except that references to the "Borrower" are to the "Grantor".

5.2 **General undertakings**

The Grantor must:

- (a) (**obligation to pay**) punctually pay the Secured Money when it becomes payable in accordance with the Transaction Documents or, in the absence of any agreement or after default under any agreement, on demand by the Secured Party;
- (b) (registration and stamping) at its own cost ensure that:
 - (i) this document is immediately registered with any Government Agency specified by the Secured Party if the Secured Party determines that registration is necessary to perfect the Security Interest granted in clause 2.1 or to protect the rights or priority of the Secured Party; and
 - (ii) this document is stamped for the proper amount within the period provided by law in each state and territory of Australia in which this document is required to be stamped; and
- (c) (ensure no Event of Default) ensure that no Event of Default occurs.

5.3 Other undertakings regarding Secured Property

The Grantor must:

(a) (outgoings)

- (i) punctually pay all outgoings (including Taxes) in respect of the Secured Property payable by it, except to the extent that:
 - (A) these are being diligently contested in good faith and by appropriate proceedings;
 - (B) it has made adequate reserves for them; and

- (C) failure to pay them will not have a Material Adverse Effect on it or prejudice the Secured Property;
- (ii) pay the contested amount after the final determination or settlement of the relevant contest; and
- (iii) on request by the Secured Party, immediately hand to the Secured Party evidence of every payment covered by this undertaking or required under this document;
- (b) (preserve and protect security) promptly do everything necessary or reasonably required by the Secured Party to:
 - (i) preserve and protect the value of the Secured Property; or
 - (ii) protect and enforce its title and rights and the Secured Party's title as secured party to the Secured Property;
- (c) (issuer information) give to the Secured Party as soon as it receives them copies of:
 - (i) each document that the Issuer gives to the Grantor in connection with the Secured Property; and
 - (ii) such other information in its possession relating to the financial condition, business, property and affairs of the Issuer as the Secured Party reasonably requests;
- (d) (comply with constitution) do everything the Grantor is required to do (including the payment of calls or instalments) by the constitution of the Issuer;
- (e) (perfection by control) to the extent that any Secured Property is of a type in which a security interest can be perfected by control under the PPSA, do anything that the Secured Party may reasonably require to enable it to perfect its Security Interest by control;
- (f) **(Sponsorship Agreement)** without limiting clause 5.3(e), ensure that:
 - (i) a Sponsorship Agreement is in force at all times in relation to all CHESS Securities in the Secured Property;
 - (ii) it complies with all its obligations under, and does not do or omit to do anything which would cause any other party to the Sponsorship Agreement to breach or to be unable to comply with its obligations under, the Sponsorship Agreement;
 - (iii) nothing is done without the consent of the Secured Party to remove any CHESS Securities from the CHESS Subregister or the control of the Controlling Participant (even if the Controlling Participant is suspended from CHESS participation); and
 - (iv) it does everything required by the Secured Party, if sponsorship of any CHESS Securities in the Secured Property is under the terms of the ASX Settlement Rules to be transferred to another Controlling Participant, to transfer that sponsorship to the new Controlling Participant;
- (g) (New Rights) give the Secured Party particulars of all New Rights and documentary or other evidence of New Rights immediately after becoming aware of the New Rights;

- (h) (take up) at the Secured Party's request take up New Rights if, in the Secured Party's opinion, failure to do so could mean the Secured Property or the Security Interest granted in clause 2.1 may become materially lessened in value or prejudicially affected;
- (i) (location of registers) ensure that:
 - (i) as at the date on which a Security Interest is granted over a Marketable Security in accordance with clause 2.1, it is recorded on the CHESS Subregister for the issuer of the Marketable Security or another register approved by the Secured Party in writing; and
 - (ii) ensure that a CHESS Security is not removed from that register without the Secured Party's consent; and

6. **SECURED PARTY'S POWERS**

6.1 Exercise of rights by Secured Party

If the Secured Party exercises a power, right, discretion or remedy in connection with this document, that exercise is taken not to be an exercise of a power, right, discretion or remedy under the PPSA unless the Secured Party states otherwise at the time of exercise. However, this clause does not apply to a power, right, discretion or remedy which can only be exercised under the PPSA.

6.2 No notice required unless mandatory

To the extent the law permits, the Grantor waives:

- (a) its rights to receive any notice that is required by:
 - (i) any provision of the PPSA (including a notice of a verification statement); or
 - (ii) any other law before a secured party or Receiver exercises a power, right, discretion or remedy; and
- (b) any time period that must otherwise lapse under any law before a secured party or Receiver exercises a power, right, discretion or remedy.

If the law which requires a period of notice or a lapse of time cannot be excluded, but the law provides that the period of notice or lapse of time may be agreed, that period or lapse is one day or the minimum period the law allows to be agreed (whichever is the longer).

However, nothing in this clause prohibits the Secured Party or any Receiver from giving a notice under the PPSA or any other law.

6.3 Secured Party's right to make good a default

- (a) If the Grantor breaches this document, the Secured Party may do everything it considers to be necessary or desirable to attempt to remedy the breach to the Secured Party's satisfaction. The Secured Party is not obliged to do so. Any liabilities or expenses incurred by the Secured Party in attempting to remedy any such breach must be reimbursed by the Grantor on demand and, pending reimbursement, will be part of the Secured Money.
- (b) Clause 6.3(a) does not limit any other right the Secured Party has under this document or at law.

6.4 Powers on enforcement

If this document has become enforceable, the Secured Party or any of its Authorised Representatives, without notice to the Grantor, may:

- (a) exercise any of the powers that might be exercised by a Receiver even if a Receiver has not been appointed; and
- (b) complete any Transfer, other transfer or instrument of any nature executed by or on behalf of the Grantor in blank and deposited with the Secured Party as Collateral Security in favour of, or take any other action required to transfer any Secured Property to, the Secured Party or any appointee of the Secured Party or any other person.

7. **POWER OF ATTORNEY**

7.1 Appointment of Attorneys

The Grantor irrevocably appoints the Secured Party and each Authorised Representative of the Secured Party and as an independent appointment appoints any Receiver, severally its attorney, at the Grantor's cost, to:

- (a) (all acts necessary) do anything necessary or desirable in the opinion of the Secured Party or the Attorney to:
 - (i) complete this document;
 - (ii) give full effect to this document;
 - (iii) better secure the Secured Property to the Secured Party in a manner consistent with this document; or
 - (iv) assist in the execution or exercise of any power under this document,

including give directions to any Controlling Participant in respect of CHESS Securities in the Secured Property, and execute any Transfer (including a Transfer in blank) or other document;

- (b) do all or any of the following while an Event of Default is continuing:
 - (i) (recover Secured Property) demand, sue for, recover and give discharge for the Secured Property;
 - (ii) (commence actions) commence, carry on , enforce, settle, arrange and compromise any proceedings to obtain or enforce the payment or delivery of the Secured Property;
 - (iii) (compound debts) compound, settle or compromise any debt of the Grantor in connection with the Secured Property;
 - (iv) (execute deeds) execute any agreement including any deed of assignment, composition or release in connection with the Secured Property;
 - (v) (exercise Marketable Security owner's rights) exercise all or any powers, rights, discretions and remedies available to any owner or holder (whether beneficial or otherwise) of any Marketable Securities included in the Secured Property (including any right to give directions to any Controlling Participant or to redeem any Marketable Security and rights

available to an owner of Marketable Securities under the Corporations Act or any other statute, or under the rules of any stock exchange); and

(vi) (**general**) do anything else that the Grantor must or may do, or that the Secured Party may do, under this document or by law.

7.2 **General**

- (a) Each Attorney may appoint and remove substitutes, and may delegate its powers (including this power of delegation) and revoke any delegation.
- (b) An Attorney may do anything contemplated by this clause even if the Attorney is affected by an actual or potential conflict of interest or duty, or might benefit from doing it.
- (c) An Attorney may do anything contemplated by this clause in its name, in the name of the Grantor or in the name of both of them.
- (d) The Grantor must ratify anything done by an Attorney under this clause.
- (e) The Grantor gives the power of attorney in this clause:
 - (i) to secure performance by the Grantor of its obligations to the Secured Party under this document and any property interest of the Secured Party under this document; and
 - (ii) for valuable consideration, receipt of which is acknowledged by the Grantor.

8. ENFORCEMENT

8.1 Circumstances when this document may be enforced

The Secured Money will immediately become payable at the Secured Party's option (despite any delay or previous waiver of the right to exercise that option) without the need for any demand or notice under this document or under another Transaction Document, and this document will immediately become enforceable (whether or not the Secured Money has become payable in this manner) if an Event of Default occurs.

8.2 Enforcement despite earlier payment

This document may be enforced:

- (a) even if the Secured Party accepts a payment of interest or other amount after the occurrence of any Event of Default; and
- (b) without the need for any notice to, or of any consent or agreement of, the Grantor or any other person.

8.3 Exclusion of PPSA provisions

To the extent the law permits:

- (a) for the purposes of sections 115(1) and 115(7) of the PPSA:
 - (i) the Secured Party need not comply with sections 95, 118, 121(4), 125, 130, 132(3)(d) or 132(4); and
 - (ii) sections 142 and 143 are excluded;

- (b) for the purposes of section 115(7) of the PPSA, the Secured Party need not comply with sections 132 and 137(3); and
- (c) if the PPSA is amended after the date of this document to permit the Grantor and the Secured Party to agree to not comply with or to exclude other provisions of the PPSA, the Secured Party may notify the Grantor that any of these provisions is excluded, or that the Secured Party need not comply with any of these provisions, as notified to the Grantor by the Secured Party.

8.4 Instructions under the Sponsorship Agreement

The Secured Party will only exercise its rights to instruct the Controlling Participant under clause 11.1 of the Sponsorship Agreement when this document becomes enforceable in accordance with clause 8.1.

9. **DEFAULT INTEREST**

9.1 Grantor must pay interest

- (a) The Grantor must pay interest on each amount that is not paid when due (unless the Grantor is already required to pay interest on the unpaid amount by the terms of any other Transaction Document), from (and including) the day on which it falls due to (but excluding) the day on which it is paid in full, at the rate calculated in accordance with clause 9.1(b). This interest must be paid on demand.
- (b) Interest on an unpaid amount accrues each day at a rate equal to the rate specified in the Facility Agreement for the payment by the Grantor of interest on unpaid amounts, and is capitalised (if not paid) on the last day of each Default Interest Period (as defined in the Facility Agreement).
- (c) This subclause does not affect the Grantor's obligation to pay each amount under this document when it is due.

9.2 Interest after judgment

If a liability of the Grantor becomes merged in a judgment or order, the Grantor, as an independent obligation, must pay interest on the amount of that liability, from (and including) the date of the judgment or order until it is paid in full, at the higher of the rate that applies under the judgment or order and the rate calculated in accordance with clause 9.1.

9.3 Accrual and calculation of interest

Interest under this clause:

- (a) accrues daily; and
- (b) is calculated on the basis of the actual number of days on which interest has accrued and of a 365 day year.

10. APPOINTMENT OF RECEIVER

10.1 Appointment

If this document has become enforceable (whether or not the Secured Party has entered into possession of all or any of the Secured Property) the Secured Party or any Authorised Representative of the Secured Party may at any time:

- (a) appoint any person or any two or more persons jointly and severally to be a receiver or receiver and manager (or an additional receiver or receiver and manager) of Secured Property;
- (b) remove the Receiver and in case of the removal, retirement or death of any Receiver appoint another as a replacement; and
- (c) fix the remuneration of the Receiver.

Subject to clause 10.2, every Receiver appointed under this subclause will be the Grantor's agent and the Grantor alone will be responsible for the Receiver's acts and defaults and remuneration.

10.2 Receiver other than as Grantor's agent

- (a) The Secured Party by notice to the Grantor and the Receiver may require the Receiver to act as the Secured Party's agent.
- (b) The power to appoint a Receiver under this clause may be exercised even though:
 - (i) at the time when this document becomes enforceable or when an appointment is made, an order may have been made or a resolution may have been passed to wind up the Grantor; or
 - (ii) a Receiver appointed in the circumstances specified in the preceding paragraph may not, or may not in some respects, act as the Grantor's agent.

10.3 Powers of Receiver

The Receiver will have full power to do all or any of the following:

- (a) (possession and control) take possession of, or take control of, collect and get any Transfers or other documents held or entitled to be held by the Secured Party under the Security Interest granted in clause 2.1 and the Secured Property and for that purpose to take proceedings (in the name of the Grantor or otherwise);
- (b) (give up possession) give up possession of any Transfers or other documents held or entitled to be held by the Secured Party under the Security Interest granted in clause 2.1 and the Secured Property;
- (c) (exercise Secured Party's rights)
 - (i) exercise all or any of the Secured Party's powers, rights, discretions and remedies under this document; and
 - (ii) comply with the directions given by the Secured Party;
- (d) (exercise Marketable Security owner's rights) exercise all or any powers, rights, discretions and remedies available to any owner or holder (whether beneficial or otherwise) of any Marketable Securities included in the Secured

Property (including any right to give directions to any Controlling Participant or to redeem any Marketable Security and rights available to an owner of Marketable Securities under the Corporations Act or any other statute, or under the rules of any stock exchange);

- (e) (**registration**) do everything necessary to obtain registration of the Secured Property in the name of the Secured Party or its nominee and to enable the Secured Party or its nominee to receive New Rights;
- (f) (New Rights) receive all New Rights;
- (q) (settle disputes)
 - (i) settle, arrange and compromise any accounts, claims, questions or disputes that may arise in connection with the Secured Property or in any way relating to this document; and
 - (ii) execute releases or other discharges in relation to the settlement, arrangement, or compromise;
- (h) (sell) sell (whether or not the Receiver has taken possession), exchange or otherwise dispose of (absolutely or conditionally) the Secured Property (or agree to do so):
 - (i) with or without any other property;
 - (ii) by public auction, private sale or tender for cash or on credit;
 - (iii) whether or not the reserve price for a sale by auction or tender is disclosed;
 - (iv) in one lot or in parcels;
 - (v) with or without special conditions, (such as conditions as to title or time or method of payment of purchase money) including by allowing the purchase money to remain:
 - (A) outstanding on any security over the property sold or over any other property; or
 - (B) owing without any security; and
 - (vi) on other terms the Receiver considers desirable,

without being responsible for any loss;

- (i) (**transfer on sale**) execute Transfers, other transfers and assignments of Secured Property (including in the name of the Grantor), and do everything to complete any sale under clause 10.3(h) that the Receiver thinks necessary;
- (j) (employees and agents) engage employees, agents, consultants, lawyers, advisers and contractors for any of the purposes of this clause on terms that the Receiver thinks appropriate;
- (k) (**give receipts**) give receipts for all money and other property that may come into the hands of the Receiver in exercise of any power given by this document;
- (enforce contracts) carry out and enforce or otherwise obtain the benefit of all contracts:

- (i) entered into or held by the Grantor in connection with the Secured Property;
- (ii) entered into in exercise of the powers given by this document;
- (make debtors bankrupt) wind up the Issuer and do everything in connection with any winding up that the Receiver thinks desirable to recover or protect Secured Property;
- (n) (perform undertakings) do everything necessary to perform any undertaking of the Grantor in this document;
- (o) (**receive money**) receive all money or other property payable or deliverable to the Grantor from the Secured Property;
- (p) (desirable or incidental matters)
 - (i) do or cause to be done everything that the Receiver thinks desirable in the interests of the Secured Party; and
 - (ii) do anything incidental to the exercise of any other power;
- (q) (take legal proceedings) take proceedings (including in the name of the Grantor) in connection with any of the above; and
- (r) (delegate) with the Secured Party's consent delegate any of the powers given to the Receiver by this clause to any person.

11. PROTECTION OF SECURED PARTY AND APPOINTEES

11.1 Protection of Secured Party and Receiver

- (a) The Secured Party is not obliged to, but may, do the following:
 - (i) notify any debtor or member of the Grantor or any other person of this document; or
 - (ii) enforce payment of any money payable to the Grantor, or take any step or proceeding for any similar purpose.
- (b) None of the Secured Party, any of its Authorised Representatives or agents, any Attorney or any Receiver is liable for any omission or delay in exercising any power, right, discretion or remedy under this document or for any involuntary loss or irregularity that may occur in relation to the exercise or non-exercise of any of them except to the extent that it is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

11.2 Conflict of interests

The Secured Party, an Authorised Representative or agent of the Secured Party, an Attorney, Receiver or other person appointed by the Secured Party under this document may exercise or agree to exercise a power given by this document or by law even though that person may have a conflict of interests in exercising the power.

11.3 Liability for loss

(a) None of the Secured Party, an Authorised Representative or agent of the Secured Party, an Attorney, a Receiver or any other person appointed by the Secured Party

under this document is liable for any loss that the Grantor suffers as a direct or indirect result of:

- the exercise or attempted exercise of, or failure to exercise, any of its rights contained in this document; and
- (ii) any release or dealing with any other Guarantee or Security Interest (including any prejudice to or loss of the Grantor's rights of subrogation),

except to the extent that such loss is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

- (b) If the Secured Party, any agent of the Secured Party or a Receiver enters into possession of Secured Property, none of the Secured Party, any of its Authorised Representatives or agents, any Attorney or any Receiver is liable:
 - (i) to account as secured party in possession or for anything except actual receipts; or
 - (ii) for any loss on realisation or for any default or omission for which a secured party in possession might be liable, except to the extent that it is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

12. **PROTECTION OF THIRD PARTIES**

12.1 **Dealings under this document**

A purchaser or other party to a disposal or dealing in attempted exercise of a power contained in this document is not:

- (a) bound to enquire whether an Event of Default has occurred, whether this document has become enforceable, whether a Receiver has been properly appointed or about the propriety or regularity of a sale, disposal or dealing; or
- (b) affected by notice that a sale, disposal or dealing is unnecessary or improper.

Despite any irregularity or impropriety in a sale, disposal or dealing, it is to be treated, for the protection of the purchaser or other party to the disposal or dealing, as being authorised by this document and valid.

12.2 Receipts

A receipt that the Secured Party, one of its Authorised Representatives or agents or a Receiver gives for any money payable to or receivable by the Secured Party or the Receiver because of this document will:

- (a) relieve the person paying or handing over money or other property from all liability:
 - (i) for the application (or any loss or misapplication) of the money or other property;
 - (ii) to enquire whether the Secured Money has become payable; and
 - (iii) (where appropriate) as to the propriety or regularity of the appointment of the Receiver; and
- (b) discharge the person paying that money from its liability to pay that money.

13. APPLICATION OF MONEY

13.1 **Order**

- (a) Money that the Secured Party or a Receiver receives under or because of this document is to be applied, after satisfaction of any claims that the Secured Party or the Receiver is aware is a claim that ranks in priority to the Security Interest granted in clause 2.1, in the following order or in any other order that the Secured Party or the Receiver may determine in its absolute discretion, subject to any Transaction Documents or applicable law to the contrary (for example, section 140 of the PPSA where relevant):
 - (i) (expenses) first in payment of all expenses that the Secured Party or a Receiver incurs in or incidental to the exercise or attempted exercise of a power or otherwise in relation to any Transaction Document;
 - (ii) (outgoings) then in payment of any other outgoings that the Receiver or the Secured Party thinks it appropriate to pay;
 - (iii) (Receiver) then in payment to the Receiver of any remuneration (whether by way of commission or otherwise);
 - (iv) (**indemnities**) then in payment to the Secured Party or a Receiver of any amount necessary to give effect to any indemnity contained in this document; and
 - (v) (Secured Money) then in payment to the Secured Party of the Secured Money.
- (b) Any surplus will belong to the Grantor or other persons entitled to it. The Secured Party or the Receiver may pay the surplus to the credit of a bank account in the name of the Grantor or other person entitled to it or into court and will then be under no further liability in relation to it. The surplus will not accrue interest.
- (c) When the Secured Party or a Receiver receives money under or because of this document, and applies it in payment to the Secured Party of the Secured Money, the Secured Party or the Receiver (as the case may be) may apply different parts of the money received to different parts of the Secured Money, regardless of any appropriation by the Grantor, as the Secured Party or Receiver chooses.

13.2 Only actual receipts credited

In applying any money towards the Secured Money, the Grantor's account will be credited only with the amount of the money that the Secured Party actually receives for that purpose. The credit will date from the time of receipt.

13.3 Compensation

If any compensation becomes payable for the Secured Property, the Secured Party may:

- (a) apply the sum received on account of any compensation, at the Secured Party's option, in or towards repayment of the Secured Money;
- (b) make, enforce, settle or compromise any claims relating to compensation; and
- (c) execute any necessary assurances and releases in the names of the Grantor and the Secured Party.

If any compensation comes into the hands of the Grantor before a final irrevocable discharge of this document, the Grantor must immediately pay it to the Secured Party.

13.4 Certificates and disputes

- (a) The Secured Party may rely on a certificate issued by any person who claims to be entitled to any money received from the exercise of any right in relation to the Secured Property which states that the Grantor owes it a certain amount of money, without making any further enquiry.
- (b) If there is any dispute between any persons (other than the Secured Party) regarding an entitlement to receive any money received from the exercise of any right in relation to the Secured Property, the Secured Party may pay that money into court, and after doing so does not have any further obligation in respect of that money.

13.5 No interest

The Secured Party is not obliged to pay interest to any person on any money received from the exercise of any right in relation to the Secured Property.

13.6 Payment into bank account

The Secured Party or the Receiver may pay any money to the credit of a bank account in the name of a person to whom it is obliged to pay any money received from the exercise of any right in relation to the Secured Property, and having done so is under no further liability in respect of that money.

13.7 Amounts contingently due

- (a) If any part of the Secured Money is contingently owing to the Secured Party when money is being applied under clause 13.1 the Secured Party or Receiver may:
 - retain an amount equal to the amount contingently owing, or any part of it;and
 - (ii) put that amount in an interest-bearing account, payable at call.
- (b) If the amount which is contingently owing:
 - (i) becomes payable; or
 - (ii) ceases to be contingently owing,

the Secured Party or Receiver must apply the amount retained (and any interest earned on it) in accordance with clause 13.1.

14. **CONTINUING SECURITY**

14.1 Continuity

The Security Interest granted in clause 2.1:

- (a) is a continuing security, and remains in full force until a final irrevocable discharge
 of that Security Interest is given to the Grantor under clause 17 despite any
 transaction or other thing (including a settlement of account or intervening
 payment); and
- (b) will apply to the present and future balance of the Secured Money.

14.2 Limitations on Grantor's rights

Until the Secured Money has been irrevocably paid and discharged in full, the Grantor may not:

- (a) share in any Guarantee, Security Interest or money received or receivable by the Secured Party in relation to the Secured Money or stand in the place of the Secured Party in relation to any Guarantee, Security Interest or right to receive money;
- (b) in reduction of its liability under this document, raise a defence, set off or counterclaim against the Secured Party or claim a set off or make a counterclaim against the Secured Party; or
- (c) claim to be entitled by way of contribution, indemnity, subrogation, marshalling or otherwise to the benefit of any document or agreement to which the Secured Party is a party.

14.3 No marshalling

The Secured Party is not under any obligation to marshal or appropriate in favour of the Grantor or to exercise, apply, perfect or recover any Security Interest that the Secured Party holds at any time or any funds or property that the Secured Party may be entitled to receive or have a claim on.

14.4 Effect of Insolvency Event

- (a) If an Insolvency Event has occurred in relation to an Obligor, any amount paid by that Obligor (relevant payment) will only be applied against any Secured Money if:
 - (i) the Secured Party forms the opinion in good faith (which will be conclusively binding on the Grantor) that it will not be required to pay the relevant payment to any person under any law relating to bankruptcy, winding up or the protection of creditors; or
 - (ii) a final judgment is given by a court of competent jurisdiction in favour of the Secured Party that it is not required to pay the relevant payment to any person under any law relating to bankruptcy, winding up or the protection of creditors.
- (b) If an amount is applied against any Secured Money and the Secured Party pays or determines that it is obliged to pay the relevant amount to any person under any law relating to bankruptcy, winding up or the protection of creditors:
 - (i) the Secured Party's rights are to be reinstated and will be the same in relation to that amount as if the application, or the payment or transaction giving rise to it, had not been made; and
 - (ii) the Grantor must immediately do anything (including the signing of documents) required by the Secured Party to restore to the Secured Party any Guarantee or Security Interest to which it was entitled immediately before that application or the payment or transaction giving rise to it.
- (c) Any discharge or release between the Secured Party and the Grantor is subject to reinstatement of the Secured Party's rights under this subclause.

15. RECOVERY OF GST

- (a) Unless otherwise indicated all amounts referred to in this document are exclusive of GST.
- (b) If the Secured Party makes a taxable supply under or in connection with this document for consideration that is exclusive of GST, the Grantor must:
 - (i) pay to the Secured Party an amount equal to any GST for which the Secured Party is liable in relation to that supply; and
 - (ii) make that payment as and when the consideration or part of it must be paid or provided.
- (c) If requested by the Grantor, the Secured Party must issue a tax invoice for a taxable supply to the person to whom it made the supply.
- (d) The Grantor's obligation to reimburse the Secured Party for an amount paid or payable to a third party (including an obligation to pay the Secured Party's or another party's legal costs) includes GST on the amount paid or payable to the third party except to the extent that the Secured Party is entitled to an input tax credit for that GST. Unless notified otherwise by the Secured Party, the Grantor must assume that the Secured Party is not entitled to any input tax credit for that GST.

16. **INDEMNITIES**

16.1 General indemnities

The Grantor must indemnify the Secured Party against, and must pay the Secured Party on demand the amount of, all losses (including loss of profit), liabilities, costs, expenses and Taxes (other than Excluded Taxes) incurred in connection with each of the following:

- (a) any Secured Money being paid or becoming due for payment other than on its due date, or any amount required to be paid under any Transaction Document not being paid on its due date, including losses (including loss of profit), liabilities, costs, expenses and Taxes (other than Excluded Taxes) incurred because of:
 - the cancellation, termination, unwinding or alteration of any swap or other arrangement made by the Secured Party to fund the Secured Money or other amount; or
 - (ii) any liquidation or re-employment of deposits or other funds acquired by the Secured Party to fund the Secured Money or other amount;
- (b) the preparation, negotiation, execution, stamping and registration of the Transaction Documents and the satisfaction of any conditions precedent;
- (c) the transactions that the Transaction Documents contemplate;
- (d) any amendment to, or any consent, approval, waiver, release or discharge of or under, a Transaction Document;
- (e) any Event of Default or Potential Event of Default;
- the administration, and any actual or attempted preservation or enforcement, of any rights under the Transaction Documents;
- (g) an enquiry by a Government Agency involving the Grantor; and

 (h) any action taken by the Secured Party under or in relation to the PPSA, including any registration, or any response to an amendment demand or a request under section 275 of the PPSA,

including legal expenses on a full indemnity basis (provided that the Grantor shall only be liable for reasonable legal expenses incurred in connection with clause 16.1(b)), the Secured Party's internal administration and legal costs at the rate and on the basis determined by the Secured Party and expenses incurred by the Secured Party in engaging investigative accountants and financial consultants (but only where such engagement is made after the occurrence of an Event of Default or a Potential Event of Default).

16.2 Indemnity for exercise of rights or proceedings

To the extent permitted by law, the Grantor must indemnify each of the Secured Party, each Authorised Representative and agent of the Secured Party, each Receiver and Attorney of the Grantor and any other person appointed under this document or the Corporations Act by or on behalf of the Secured Party as secured party under this document against, and must pay each of them on demand the amount of all losses, liabilities, costs, expenses and Taxes (other than Excluded Taxes) that they each incur:

- (a) (directly or indirectly) in the exercise or attempted exercise of any of the powers, rights, discretions or remedies (express or implied) vested in them under this document or the Corporations Act; and
- (b) in connection with all proceedings, expenses, claims and demands in relation to anything done or omitted in any way relating to Secured Property,

including legal expenses on a full indemnity basis and expenses incurred in engaging investigative accountants and financial consultants (but only where such engagement is made after the occurrence of an Event of Default or a Potential Event of Default) except to the extent that any such losses, liabilities, costs, expenses and Taxes are a direct and foreseeable result of the fraud, gross negligence or wilful misconduct of that person.

16.3 **Currency indemnity**

If, for any reason (including as a result of a judgment or order), an amount payable by the Grantor under or in respect of this document (**Relevant Amount**) is received by the Secured Party in a currency (**Payment Currency**) that is not the currency in which the amount is expressed to be payable under this document (**Required Currency**) then the Grantor, as an independent obligation, must indemnify the Secured Party against, and must pay the Secured Party on demand the amount of, any shortfall between:

- (a) the amount of Required Currency which the Secured Party receives on converting the amount it received in the Payment Currency into an amount in the Required Currency in accordance with its usual practice; and
- (b) the Relevant Amount in the Required Currency.

16.4 Recovery from Secured Property

A person who is entitled to be indemnified for a loss, liability, expense or Tax under clauses 16.1, 16.2 or 16.3 may recover the amount to be indemnified direct from the Secured Property.

17. **DISCHARGE**

The Secured Party must at the request and cost of the Grantor reconvey, surrender or release any remaining Secured Property (as appropriate) to the Grantor and the Secured Property will then be discharged from this document:

- (a) when the Secured Party is satisfied that:
 - (i) all the Secured Money has been irrevocably paid and discharged in full; and
 - (ii) no amount remains contingently payable or may become payable on the security of this document (including under an indemnity); and
- (b) on payment or retention of all expenses incurred by or payable to the Secured Party, its Authorised Representatives or any Receiver or Attorney.

Any discharge is subject to clause 14.4.

18. **CONFIDENTIALITY**

18.1 General confidentiality

The parties to this document agree that the Secured Party must not disclose any information provided by the Grantor under the Transaction Documents to any person who is not a party, and the Grantor must not disclose any information concerning the contents of or the transactions contemplated by any Transaction Document, including in each case information of the kind mentioned in section 275(1) of the PPSA, to any person who is not a party, except to the extent that:

- (a) (**permitted by documents**) the disclosure is expressly permitted by a Transaction Document or required by law to ensure the enforceability or priority of any Transaction Document;
- (b) (consent of other party) the other party consents to the disclosure;
- (c) (**public domain**) the information is already in the public domain, unless it entered the public domain because of a breach of confidentiality by the disclosing party;
- (d) (employees and advisers) the disclosure is made on a confidential basis to the officers, employees, agents, lawyers, auditors or other professional advisers of the party or its holding company. For the avoidance of doubt, it is not mandatory for a receiving party to enter into any written confidentiality undertaking in respect of any disclosure made under this paragraph;
- (e) (comply with laws) the disclosure is necessary to comply with any applicable law, or an order of a court or tribunal;
- (f) (**comply with directives**) the disclosure is necessary to comply with a directive or request of any Government Agency (whether or not having the force of law);
- (g) (**obtain Authorisations**) the disclosure is necessary or desirable to obtain an Authorisation from any Government Agency;
- (h) (discovery and litigation) the disclosure is necessary or desirable in relation to any discovery of documents, or any proceedings before a court, tribunal or other Government Agency; or
- (i) (**Secured Party's purposes**) it is in the Secured Party's interest to disclose the information, including:

- (i) to persons considering purchasing the Secured Party or who are involved in valuing the Secured Party for ratings purposes; or
- (ii) to persons considering taking an assignment of, interest in or exposure to the Facility or the Transaction Documents, including under securitisation or sub-participation arrangements.

Any disclosure made under this clause 18.1(i) must be made on the basis that the receiving party must keep the information confidential as required by this clause.

18.2 No request

The Grantor agrees not to exercise its rights to make any request of the Secured Party under section 275 of the PPSA, to authorise the disclosure of any information under that section or to waive any duty of confidence that would otherwise permit non-disclosure under that section (but this does not limit the Grantor's rights to request information other than under section 275).

19. **NOTICES**

19.1 How to give a notice

A notice, consent or other communication under this document is only effective if it is:

- (a) in writing, signed by or on behalf of the person giving it;
- (b) addressed to the person to whom it is to be given; and
- (c) given in one of the following ways:
 - (i) sent by prepaid mail (by airmail, if the addressee is overseas) or delivered to that person's address;
 - (ii) sent by fax to that person's fax number and the machine from which it is sent produces a report that states that it was sent in full without error;
 - (iii) given personally; or
 - (iv) given in any other manner permitted by law.

19.2 When a notice is given

Subject to clause 19.3, a notice, consent or other communication that complies with this clause is conclusively regarded as given and received:

- (a) if it is sent by fax or delivered, if received:
 - (i) by 5.00 pm (local time in the place of receipt) on a Business Day on that day; or
 - (ii) after 5.00 pm (local time in the place of receipt) on a Business Day, or on a day that is not a Business Day – on the next Business Day;
- (b) if it is sent by mail, when it would be delivered in the ordinary course of post, but in any event:
 - (i) not later than three Business Days after posting within Australia; or

- (ii) not later than seven Business Days after posting to or from a place outside Australia;
- (c) if given personally, when actually received by that person; and
- (d) if it is given in any other manner permitted by law, when actually received by that person, unless a later time of receipt is specified in it.

19.3 Notices to the Secured Party

Any notice, consent or other communication to be made or delivered to the Secured Party will be effective:

- (a) only when actually received by it; and
- (b) only if it is expressly marked for the attention of the department or officer specified in the following clause (or any substitute department or officer as it specifies for this purpose).

19.4 Address for notices

A person's address and fax number are those set out below, or as the person notifies the sender:

Secured Party

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61 2 9324 3170

Attention: Justin Churchill / Luke Goldsworthy

Email: justin.churchill@ubs.com / luke.goldsworthy@ubs.com

Grantor

Address: Department of Treasury

P O Box 542 Waigani 131 N.C.D.

Papua New Guinea

Fax number: +675 3128804

Attention: Mr Dairi Vele, Secretary for Department of Treasury

Email address: dairi_vele@treasury.gov.pg

Telephone: +675 3235600

20. AMENDMENT AND ASSIGNMENT

20.1 Amendment

This document can only be amended or replaced by another document executed by the parties.

20.2 Assignment

(a) The Grantor may only assign, encumber, declare a trust over or otherwise deal with its rights under any Transaction Document with the written consent of the Secured Party.

- (b) Subject to the terms of the Security Trust Deed, the Secured Party may assign, encumber, declare a trust over or otherwise deal with its rights under this document without the consent of the Grantor, and may disclose to any potential holder of the right, or an interest in the right, any information relating to this document or any party to it.
- (c) After any disposal by the Secured Party of its rights under the Security Interest granted in clause 2.1, the Secured Party may serve a Participant Change Notice on the Grantor and the Grantor must agree to the novation of the Sponsorship Agreement to a Controlling Participant acceptable to the Secured Party.

21. GENERAL

21.1 Governing law

- (a) This document is governed by the laws of the State of New South Wales.
- (b) Each party submits to the non-exclusive jurisdiction of the courts of the State of New South Wales, and any court that may hear appeals from any of those courts, for any proceedings in connection with any Transaction Document.
- (c) The Grantor irrevocably waives:
 - (i) any objection to the venue of any proceedings on the ground that they have been brought in an inconvenient forum; and
 - (ii) any immunity from set off, suits, proceedings and execution to which it or any of its property may now or in the future be entitled under any applicable law.
- (d) The Grantor confirms the appointment of NOROSE NOTICES AUSTRALIA PTY LTD ACN 158 029 586 under clause 28 of the Facility Agreement of Level 16, Grosvenor Place, 225 George Street, Sydney, NSW 2000 as its agent to receive service of process for any proceedings in connection with any Transaction Document. The Grantor undertakes to maintain this appointment until all Secured Money is repaid, and agrees that any process served on that person is taken to be served on it.

21.2 Liability for own expenses

The Grantor is liable for its own costs and expenses in complying with this document, including where it does so at the Secured Party's request or for the Secured Party's benefit.

21.3 Giving effect to this document

- (a) The Grantor must do anything (including executing any Transfer and other transfer in blank, or any other document, and perfecting and protecting any Security Interest intended to be created by or pursuant to this document), and must ensure that its employees and agents do anything, that the Secured Party may reasonably require to:
 - (i) give full effect to this document; or
 - (ii) more fully secure the rights, remedies and powers of the Secured Party under this document or to enable the Secured Party to exercise those rights, remedies and powers,

including giving directions to any Controlling Participant in respect of any CHESS Securities in the Secured Property.

(b) The Secured Party may, at the Grantor's cost, do anything which the Grantor should have done under this document if the Grantor does not do so promptly or, if in the Secured Party's opinion, the Grantor does not do so properly.

21.4 Authority to register and waiver of right to receive verification statements

The Grantor acknowledges that the Secured Party may, at the Grantor's cost, register one or more financing statements in relation to its Security Interest. If permitted by the PPSA, the Grantor waives its right under section 157 of the PPSA to receive notice of any verification statement relating to the registration of any such financing statement or any related financing change statement.

21.5 Variation of rights

The exercise of a right partially or on one occasion does not prevent any further exercise of that right in accordance with the terms of this document. Neither a forbearance to exercise a right nor a delay in the exercise of a right operates as an election between rights or a variation of the terms of this document.

21.6 **Operation of this document**

- (a) Subject to clause 21.6(b), the Transaction Documents contain the entire agreement between the parties about their subject matter. Any previous understanding, agreement, representation or warranty relating to that subject matter is replaced by the Transaction Documents and has no further effect.
- (b) Any right that the Secured Party may have under the Transaction Documents is in addition to, and does not replace or limit, any other right that the Secured Party may have.
- (c) Any provision of this document which is unenforceable or partly unenforceable is, where possible, to be severed to the extent necessary to make this document enforceable, unless this would materially change the intended effect of this document.

21.7 Operation of indemnities

- (a) Each indemnity in this document survives the expiry or termination of this document.
- (b) The Secured Party may recover a payment under an indemnity in this document before it makes the payment in respect of which the indemnity is given.
- (c) If a provision of this document is expressed to:
 - (i) indemnify;
 - (ii) exclude or limit any liability of; or
 - (iii) otherwise benefit,

a person who is not a party to this document, the Grantor agrees that the Secured Party holds the benefit of that indemnity, exclusion, limitation or other benefit for that person and may enforce this document on their behalf and for their benefit.

21.8 Consents

Where this document contemplates that the Secured Party may agree or consent to something (however it is described), the Secured Party may:

- (a) agree or consent, or not agree or consent, in its absolute discretion; and
- (b) agree or consent subject to conditions,

unless this document expressly contemplates otherwise.

21.9 Statements by Secured Party

A statement by an Authorised Representative of the Secured Party on any matter relating to any Transaction Document (including any amount owing by the Grantor) is conclusive unless clearly wrong on its face.

21.10 Set-off

If an Event of Default occurs and is continuing, the Secured Party, without notice to the Grantor, may combine any account that the Grantor holds at any branch or office (in Australia or elsewhere) of the Secured Party with, or set off any amount in any currency that is or may become owing in any currency by the Secured Party to the Grantor against, any amount owing by the Grantor to the Secured Party under the Transaction Documents. For this purpose the Secured Party may:

- (a) change the terms (including the repayment date) of any account or other payment obligation between the parties;
- (b) convert amounts into different currencies in accordance with the Secured Party's usual practice; and
- (c) do anything (including execute any document) in the name of the Grantor that the Secured Party considers necessary or desirable.

This subclause overrides any other document or agreement to the contrary.

21.11 No merger

Nothing in this document merges with any other Security Interest, or any Guarantee, judgment or other right or remedy, that the Secured Party may hold at any time.

21.12 Exclusion of contrary legislation

Any legislation that affects an obligation of the Grantor in a manner that is adverse to the interests of the Secured Party, or adversely affects the exercise by the Secured Party of a right or remedy, under or relating to this document is excluded to the full extent permitted by law.

21.13 Counterparts

This document may be executed in counterparts. Delivery of a counterpart of this document by email attachment or fax constitutes an effective mode of delivery.

21.14 Execution by fewer than all parties

This document binds the Grantor even if the Secured Party does not execute or only subsequently executes this document.

SCHEDULE 1

Initial Securities

1. CHESS SECURITIES

Name and ARBN of Issuer: Oil Search Limited, ARBN 055 079 868

Description of Marketable Securities: Fully Paid Ordinary Shares

Number of Marketable Securities: 12,377,994

Holder Identification or Shareholder Reference Number: The number notified by UBS

Nominees Pty Ltd to the other party to this document.

EXECUTED as a deed.

Each person who executes this document on behalf of a party under a power of attorney declares that he or she is not aware of any fact or circumstance that might affect his or her authority to do so under that power of attorney.

Agreed and accepted for and on behalf of the Independent State of Papua New Guinea:

Signed, Sealed and Delivered for and on behalf of the **INDEPENDENT STATE OF PAPUA NEW GUINEA** by Sir Michael Oglo, G.C.M.G., K.St.J., Governor General, acting with and in accordance with the advice of the National Executive Council:

Ula

in the presence of:

Name: CARL OKUK

Title:

Signed, Sealed and Delivered for UBS Nominees Pty Ltd under power of attorney in the presence of:

Witness Signature

KEVIN LU

Print Name

Signature of attorney

Kevin Bush Managing Director

Print Name

signature of attorne

Lance McKegg Executive Director

Print Name

Specific Security Deed (CHESS Securities - Collar)

Independent State of Papua New Guinea

and

UBS AG, Australia Branch

ABN 47 088 129 613

Project Kumul

/2 March 2014

CONTENTS

CLAUSE					
1.	INTER	PRETATION4			
	1.1 1.2 1.3 1.4 1.5	Definitions			
2.	SECU	RITY8			
	2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8 2.9	Security clause			
3.	DEALI	NGS WITH SECURED PROPERTY10			
	3.1	Restricted dealings10			
4.	REPRE	SENTATIONS AND WARRANTIES10			
	4.1 4.2 4.3 4.4	General representations and warranties of the Grantor			
5.	GRANT	OR'S UNDERTAKINGS11			
	5.1 5.2 5.3	Undertakings of the Borrower 11 General undertakings 11 Other undertakings regarding Secured Property 12			
6.	SECUR	SECURED PARTY'S POWERS			
	6.1 6.2 6.3 6.4	Exercise of rights by Secured Party			
7.	POWER	POWER OF ATTORNEY			
	7.1 7.2	Appointment of Attorneys			
8.	ENFOR	CEMENT			
	8.1 8.2 8.3	Circumstances when this document may be enforced			
9.	DEFAULT INTEREST				
	9.1 9.2 9.3	Grantor must pay interest			
10.	APPOINTMENT OF RECEIVER				

	10.1 10.2 10.3	Appointment			
11,	PROTE	ECTION OF SECURED PARTY AND APPOINTEES20			
	11.1 11.2 11.3	Protection of Secured Party and Receiver			
12.	PROTECTION OF THIRD PARTIES				
	12.1 12.2	Dealings under this document			
13.	APPLICATION OF MONEY21				
	13.1 13.2 13.3 13.4 13.5 13.6 13.7	Order21Only actual receipts credited22Compensation22Certificates and disputes22No interest22Payment into bank account23Amounts contingently due23			
14.	CONTINUING SECURITY				
	14.1 14.2 14.3 14.4	Continuity23Limitations on Grantor's rights23No marshalling24Effect of Insolvency Event24			
15.	RECOV	/ERY OF GST24			
16.	INDEMNITIES				
	16.1 16.2 16.3 16.4	General indemnities25Indemnity for exercise of rights or proceedings25Currency indemnity26Recovery from Secured Property26			
17.	DISCH	ARGE			
18.	CONFIDENTIALITY27				
	18.1 18.2	General confidentiality			
19.	NOTIC	ES28			
	19.1 19.2 19.3 19.4	How to give a notice			
20.	AMENDMENT AND ASSIGNMENT29				
	20.1 20.2	Amendment 29 Assignment 29			
21.	GENERAL				
	21.1 21.2 21.3 21.4 21.5 21.6	Governing law			

21.7	Operation of indemnities	31
21.8	Consents	31
21.9	Statements by Secured Party	32
21.10	Set-off	32
	No merger	
21.12	Exclusion of contrary legislation	32
	Counterparts	
21.14	Execution by fewer than all parties	32

THIS DEED is made on /2 March 2014

BETWEEN:

- (1) Independent State of Papua New Guinea (the Grantor); and
- (2) **UBS AG, Australia Branch** ABN 47 088 129 613 of Level 16, Chifley Tower, 2 Chifley Place, Sydney NSW 2000, Australia (the **Secured Party**).

RECITALS

- (A) The parties entered into the Equity Derivative Financing Agreement on or around the date of this document.
- (B) The parties enter into this document in connection with the Equity Derivative Financing Agreement.

THE PARTIES AGREE AS FOLLOWS:

1. INTERPRETATION

1.1 Definitions

The following definitions apply in this document.

ASX Settlement means ASX Settlement Pty Limited (ABN 49 008 504 532).

ASX Settlement Rules means the ASX Settlement Operating Rules and any other operating rules, procedures, directions, decisions, requirements, customs, usages and practices of ASX Settlement, as amended from time to time.

Attorney means an attorney appointed under an Equity Derivative Financing Document and any attorney's substitute or delegate.

CHESS Security means a Marketable Security that is registered on a CHESS Subregister.

CHESS Subregister means, for the Issuer, that part of its share register that is maintained by ASX Settlement under CHESS.

Collateral Security means a Security Interest or Guarantee (other than the Security Interest granted in clause 2.1) from any person that secures or otherwise provides for payment of any Secured Money.

Controlling Participant means UBS Securities Australia Limited ABN 62 008 586 481 or any other person acceptable to the Secured Party who is able to sponsor Marketable Securities on a CHESS Subregister.

Event of Default means an "Event of Default" (as defined in the Equity Derivative Financing Agreement) in respect of which the Grantor is the Defaulting Party or a "Termination Event" (as defined in the Equity Derivative Financing Agreement) in respect of which the Grantor is the Affected Party.

Equity Derivative Financing Agreement means the confirmation letter dated on or about the date of this document between the Grantor and the Secured Party, including any ISDA master agreement or credit support annex incorporated in or entered into under or in relation to the letter or the transactions contemplated therein.

Issuer means Oil Search Limited, a company incorporated under the laws of the Independent State of Papua New Guinea with ARBN 055 079 868.

Marketable Security means:

- (a) an intermediated security;
- (b) a debenture, stock or bond, unit in a unit trust, or other marketable security; and
- (c) any other investment instrument.

New Right means a present or future right of the Grantor:

- (a) to or in any money, dividend (including any return of capital), interest, offer, bonus, note or other Marketable Security, or any entitlement to subscribe for any of them;
- (b) resulting from any substitution, conversion, redemption, forfeiture, cancellation, reclassification, consolidation or subdivision; or
- (c) resulting from a reduction of capital, liquidation or scheme of arrangement,

in connection with the Secured Property.

Nominee means UBS Nominees Pty Ltd ABN 32 001 450 522.

Nominee Deed means the Nominee Deed dated on or about the date of this document between the Grantor, the Secured Party, the Nominee and the Controlling Participant.

Permitted Security Interest means:

- (a) a Security Interest (if any) created under an Equity Derivative Financing Document;
- a lien that arises by operation of law in the ordinary course of ordinary business, where the amount secured is not overdue or is being diligently contested in good faith and appropriately provisioned;
- (c) a Security Interest:
 - (i) existing on the date of this document that has been approved by the Secured Party; or
 - (ii) that arises after the date of this document and that the Secured Party approves before it arises,

where the amount secured does not increase, and the time for payment of that amount is not extended beyond the amount and time approved by the Secured Party.

Receiver means a receiver or a receiver and manager.

Secured Money means all amounts (including damages) that are payable, owing but not payable, or that otherwise remain unpaid by the Grantor to the Secured Party and the Finance Parties on any account at any time under or in connection with the Equity Derivative Financing Agreement or any other Equity Derivative Financing Document:

- (a) whether present or future, actual or contingent;
- (b) whether incurred alone, jointly, severally or jointly and severally;

- (c) whether the Grantor is liable on its own account or for the account of, or as surety for, another person and without regard to the capacity in which the Grantor is liable;
- (d) whether due to the Secured Party alone or with another person;
- (e) whether the Secured Party is entitled for its own account or the account of another person;
- (f) whether arising from a banker and customer relationship or any other relationship;
- (g) whether originally contemplated by the Grantor or the Secured Party or not;
- (h) whether the Secured Party is the original person in whose favour the undertakings in this document, the Equity Derivative Financing Agreement or any other Equity Derivative Financing Document were given or an assignee and, if the Secured Party is an assignee:
 - (i) whether or not the Grantor consented to or knew of the assignment;
 - (ii) no matter when the assignment occurred; and
 - (iii) whether or not the entitlements of that original person were assigned with the Security Interest granted in clause 2.1; and
- (i) if determined pursuant to any award, order or judgment against the Grantor, whether or not the Grantor was a party to the court proceedings, arbitration or other dispute resolution process in which that award, order or judgment was made.

Secured Property means all the Grantor's present and future interest in, to, under or derived from all:

- (a) Transferred Securities; and
- (b) New Rights.

Transfer means, in respect of a Marketable Security or a New Right, an executed document of transfer sufficient to transfer all the legal and beneficial ownership of that Marketable Security or New Right to the Secured Party or its nominee.

Transferred Security means, at any time, each Marketable Security of the Issuer held by, the Nominee as trustee for the Grantor pursuant to the Nominee Deed at that time.

1.2 Definitions in ASX Settlement Rules

Unless otherwise defined in this document, any capitalised term used in this document which is defined in the:

- (a) ASX Settlement Rules has the meaning given in the ASX Settlement Rules; and
- (b) an Equity Derivative Financing Agreement, including by incorporation, has the same meaning given in the Equity Derivative Financing Agreement.

1.3 Rules for interpreting this document

Headings are for convenience only, and do not affect interpretation. The following rules also apply in interpreting this document, except where the context makes it clear that a rule is not intended to apply.

(a) A reference to:

- (i) a legislative provision or legislation (including subordinate legislation) is to that provision or legislation as amended, re-enacted or replaced, and includes any subordinate legislation issued under it;
- (ii) a document (including this document) or agreement, or a provision of a document (including this document) or agreement, is to that document, agreement or provision as amended, supplemented, replaced or novated;
- (iii) a party to this document or to any other document or agreement includes a successor in title, permitted substitute or a permitted assign of that party;
- (iv) a person includes any type of entity or body of persons, whether or not it is incorporated or has a separate legal identity, and any executor, administrator or successor in law of the person; and
- (v) anything (including a right, obligation or concept) includes each part of it and any part of it.
- (b) A singular word includes the plural, and vice versa.
- (c) A word which suggests one gender includes the other genders.
- (d) If a word or phrase is defined, any other grammatical form of that word or phrase has a corresponding meaning.
- (e) If an example is given of anything (including a right, obligation or concept), such as by saying it includes something else, the example does not limit the scope of that thing.
- (f) The word **agreement** includes an undertaking or other binding arrangement or understanding, whether or not in writing.
- (g) The expression **this document** includes the agreement, arrangement, understanding or transaction recorded in this document.
- (h) An Event of Default "continues" until the Secured Party notifies the Grantor that it has been:
 - (i) remedied to the satisfaction of the Secured Party; or
 - (ii) waived by the Secured Party.
- (i) Unless otherwise defined in this document, words that are defined in an applicable Accounting Standard have the same meaning in this document.
- (j) Unless otherwise defined in this document, words defined in the GST Law have the same meaning in clauses concerning GST.
- (k) If a person is a member of a GST group, references to GST for which the person is liable and to input tax credits to which the person is entitled include GST which the representative member of the GST group is liable and input tax credits to which the representative member is entitled.

1.4 PPSA terms

In this document, unless the context requires otherwise, the following terms have the meanings given to them in the PPSA:

- (a) amendment demand;
- (b) attach;
- (c) control;
- (d) financing change statement;
- (e) financing statement;
- (f) intermediated security;
- (g) investment instrument;
- (h) registration; and
- (i) verification statement.

1.5 Non Business Days

If the day on or by which a person must do something under this document is not a Business Day:

- (a) if the act involves a payment that is due on demand, the person must do it on or by the next Business Day; and
- (b) in any other case, the person must do it on or by the previous Business Day.

2. **SECURITY**

2.1 Security clause

The Grantor grants a security interest in the Secured Property to the Secured Party to secure the payment of the Secured Money. This security interest is a fixed charge.

2.2 **Priority**

The Security Interest granted in clause 2.1 is intended to take effect as a first ranking security subject only to those Permitted Security Interests which the Secured Party agrees in writing rank in priority to it.

2.3 Transferred Securities

The Grantor must:

- (a) at the time it executes this document, enter into the Nominee Deed with the Secured Party, the Nominee and the Controlling Participant in respect of the Transferred Securities; and
- (b) at each time that any transfer of Marketable Securities of the Issuer from the Secured Party to the Nominee occurs under the Nominee Deed:
 - give the Secured Party, the Nominee and the Controlling Participant the applicable holder or other identification numbers and information sufficient to identify the Transferred Securities (including a copy of the relevant holding statement);
 - (ii) do everything within its power that is necessary to ensure that the relevant Transferred Securities are recorded on the Issuer's CHESS Subregister; and

(iii) do everything within its power to transfer sponsorship of the relevant Transferred Securities to the Controlling Participant.

This clause does not apply to the extent that the Secured Party agrees otherwise under clause 2.9.

2.4 Documents relating to New Rights

The Grantor must deposit with the Secured Party or the Secured Party's nominee:

- (a) all certificates, transfers and other documents or agreements evidencing title to New Rights; and
- (b) Transfers (in form and number satisfactory to the Secured Party) with the name of the transferee, the consideration and the date left blank, in respect of those new Rights,

immediately it acquires those New Rights.

2.5 Dividends and votes

The Grantor may exercise rights (including voting rights) in connection with the Secured Property to the extent contemplated by the Nominee Deed.

2.6 Secured Party entitled to exercise rights on default

If an Event of Default occurs and is continuing:

- (a) the Grantor must, if the Secured Party gives notice to the Grantor requiring it to do so, do everything necessary to ensure that the Secured Property is registered in the name of the Secured Party or its nominee in accordance with any directions contained in that notice;
- (b) the Grantor's rights under clause 3.2 (Voting) of the Nominee Deed immediately cease;
- (c) the Secured Party may exercise or refrain from exercising any rights in connection with the Secured Property; and
- (d) the Secured Party may receive all New Rights and apply them (or their sale proceeds) in accordance with clause 13.1.

2.7 Acknowledgment of no subordination

The Grantor acknowledges that the Secured Party has not agreed to subordinate its Security Interest in the Secured Property to any other interest in the Secured Property, except to the extent (if any) expressly provided by an Equity Derivative Financing Document.

2.8 Collateral Security

The Security Interest granted in clause 2.1 is collateral to and secures the same money as is secured by Collateral Securities.

2.9 Nominees and custodians

If the Secured Party agrees that any of the Secured Property can be held by a nominee or custodian on behalf of the Grantor other than the Nominee, the Grantor undertakes to procure that the nominee or custodian:

- (a) acknowledges and consents to the Security Interest created by clause 2.1;
- (b) agrees to hold the Secured Property on such terms as the Secured Party requires, including by agreeing not to:
 - (i) create or allow another interest in any Secured Property other than a Permitted Security Interest; or
 - (ii) dispose, or part with possession, of any Secured Property,

without the consent of the Secured Party; and

(c) agrees to do anything, including executing documents, that the Secured Party may require to enable it to perfect its Security Interest by control.

3. **DEALINGS WITH SECURED PROPERTY**

3.1 Restricted dealings

The Grantor must not do, or agree to do, any of the following unless it is permitted to do so by another provision in an Equity Derivative Financing Document:

- (a) create or allow another interest in any Secured Property other than a Permitted Security Interest; or
- (b) dispose, or part with possession, of any Secured Property.

4. REPRESENTATIONS AND WARRANTIES

4.1 General representations and warranties of the Grantor

The provisions of paragraph 1 (*Disclosure requirement and additional representations and warranties*) of the Equity Derivative Financing Agreement apply to this document as if set out in full in this document, except that references to "Party B" are to the "Grantor", references to "Party A" are to the "Secured Party" and with any other necessary changes.

4.2 Representations and warranties regarding Secured Property

The Grantor represents and warrants to the Secured Party that:

(a) (Secured Property)

- (i) it is the beneficial owner of the Secured Property (and the Secured Property is held by the Nominee in accordance with the Nominee Deed), and
- (ii) it has full power to grant a Security Interest in the Secured Property in the manner provided in this document; and
- (iii) the Security Interest granted in clause 2.1 is an effective security in the Secured Property, except to the extent (if any) set out in any Equity Derivative Financing Document or that the Secured Party may otherwise agree;
- (b) (no other interest) no other person has any interest in or other right over the Secured Property except:
 - (i) the Nominee, pursuant to the terms of the Nominee Deed;

- (ii) to the extent (if any) set out in any other Equity Derivative Financing Document;
- (iii) Permitted Security Interests; or
- (iv) as otherwise agreed by the Secured Party;
- (c) (**location of Secured Property**) at the date of execution of this document, none of the Secured Property is located, or taken for the purposes of New South Wales stamp duty law to be located, in New South Wales;
- (d) (no escrow or other conditions) the Secured Property is not subject to any escrow or other conditions imposed by the Corporations Act or under the rules of any stock exchange;
- (e) (all calls satisfied) all calls made in respect of the Secured Property have been satisfied;
- (f) (fully paid Marketable Securities) the Marketable Securities in the Secured Property are all fully paid; and
- (g) (**no money owing to Issuer**) no money is owing by the Grantor to the Issuer in connection with the Secured Property.

4.3 Repetition of representations and warranties

The representations and warranties in:

- (a) clause 4.1 (General representation and warranties of the Grantor) are taken to be repeated on each date that the corresponding representation or warranty in the Equity Derivative Financing Agreement is repeated under the terms of the Equity Derivative Financing Agreement, on the basis of the facts and circumstances as at that date; and
- (b) clause 4.2 (*Representations and warranties regarding Secured Property*) are taken to be repeated on each date from the date of this document until the date on which all amounts outstanding under the Equity Derivative Financing Documents have been paid in full, on the basis of the facts and circumstances as at that date.

4.4 No representations by the Secured Party

The Grantor acknowledges that it has not relied and will not rely on any financial or other advice, representation, statement or promise provided or made by or on behalf of the Secured Party in deciding to enter into this document or to exercise any right or perform any obligation under it.

5. GRANTOR'S UNDERTAKINGS

5.1 Undertakings of the Borrower

Clause 13 of the Bridge Facility Agreement is incorporated into this document as if set out in full, except that references to the "Borrower" are to the "Grantor", references to "this document" or "Transaction Documents" are to the "Equity Derivative Financing Documents", and with any other necessary changes.

5.2 General undertakings

The Grantor must:

- (a) (obligation to pay) punctually pay the Secured Money when it becomes payable in accordance with the Equity Derivative Financing Documents or, in the absence of any agreement or after default under any agreement, on demand by the Secured Party;
- (b) (registration and stamping) at its own cost ensure that:
 - (i) this document is immediately registered with any Government Agency specified by the Secured Party if the Secured Party determines that registration is necessary to perfect the Security Interest granted in clause 2.1 or to protect the rights or priority of the Secured Party; and
 - (ii) this document is stamped for the proper amount within the period provided by law in each state and territory of Australia in which this document is required to be stamped; and
- (c) (ensure no Event of Default) ensure that no Event of Default in respect of which the Grantor is the Defaulting Party occurs.

5.3 Other undertakings regarding Secured Property

The Grantor must:

(a) (outgoings)

- (i) punctually pay all outgoings (including Taxes) in respect of the Secured Property payable by it, except to the extent that:
 - (A) these are being diligently contested in good faith and by appropriate proceedings;
 - (B) it has made adequate reserves for them; and
 - failure to pay them will not have a Material Adverse Effect on it or prejudice the Secured Property;
- (ii) pay the contested amount after the final determination or settlement of the relevant contest; and
- (iii) on request by the Secured Party, immediately hand to the Secured Party evidence of every payment covered by this undertaking or required under this document:
- (b) (preserve and protect security) promptly do everything necessary or reasonably required by the Secured Party to:
 - (i) preserve and protect the value of the Secured Property; or
 - (ii) protect and enforce its title and rights and the Secured Party's title as secured party to the Secured Property;
- (c) (issuer information) give to the Secured Party as soon as it receives them copies of:
 - (i) each document that the Issuer gives to the Grantor in connection with the Secured Property; and

- such other information in its possession relating to the financial condition, business, property and affairs of the Issuer as the Secured Party reasonably requests;
- (d) (comply with constitution) do everything the Grantor is required to do (including the payment of calls or instalments) by the constitution of the Issuer;
- (e) (perfection by control) to the extent that any Secured Property is of a type in which a security interest can be perfected by control under the PPSA, do anything that the Secured Party may reasonably require to enable it to perfect its Security Interest by control;
- (f) (Nominee Deed) without limiting clause 5.3(e), ensure that:
 - the Nominee Deed is in force at all times in relation to all CHESS Securities in the Secured Property;
 - (ii) it complies with all its obligations under, and does not do or omit to do anything which would cause any other party to the Nominee Deed to breach or to be unable to comply with its obligations under, the Nominee Deed;
 - (iii) it does nothing, and it does not instruct or otherwise cause the Nominee to do anything, without the consent of the Secured Party to remove any CHESS Securities from the CHESS Subregister or the control of the Controlling Participant (even if the Controlling Participant is suspended from CHESS participation); and
 - (iv) it does everything required by the Secured Party, if sponsorship of any CHESS Securities in the Secured Property is under the terms of the ASX Settlement Rules to be transferred to another Controlling Participant, to transfer that sponsorship to the new Controlling Participant;
- (g) (New Rights) give the Secured Party particulars of all New Rights and documentary or other evidence of New Rights immediately after becoming aware of the New Rights;
- (h) (take up) at the Secured Party's request take up New Rights if, in the Secured Party's opinion, failure to do so could mean the Secured Property or the Security Interest granted in clause 2.1 may become materially lessened in value or prejudicially affected; and
- (i) (**location of registers**) ensure that:
 - (i) as at the date on which a Security Interest is granted over a Marketable Security in accordance with clause 2.1, it is recorded on the CHESS Subregister for the issuer of the Marketable Security or another register approved by the Secured Party in writing; and
 - (ii) ensure that a CHESS Security is not removed from that register without the Secured Party's consent.

6. SECURED PARTY'S POWERS

6.1 Exercise of rights by Secured Party

If the Secured Party exercises a power, right, discretion or remedy in connection with this document, that exercise is taken not to be an exercise of a power, right, discretion or remedy under the PPSA unless the Secured Party states otherwise at the time of exercise.

However, this clause does not apply to a power, right, discretion or remedy which can only be exercised under the PPSA.

6.2 No notice required unless mandatory

To the extent the law permits, the Grantor waives:

- (a) its rights to receive any notice that is required by:
 - (i) any provision of the PPSA (including a notice of a verification statement); or
 - (ii) any other law before a secured party or Receiver exercises a power, right, discretion or remedy; and
- (b) any time period that must otherwise lapse under any law before a secured party or Receiver exercises a power, right, discretion or remedy.

If the law which requires a period of notice or a lapse of time cannot be excluded, but the law provides that the period of notice or lapse of time may be agreed, that period or lapse is one day or the minimum period the law allows to be agreed (whichever is the longer).

However, nothing in this clause prohibits the Secured Party or any Receiver from giving a notice under the PPSA or any other law.

6.3 **Secured Party's right to make good a default**

- (a) If the Grantor breaches this document, the Secured Party may do everything it considers to be necessary or desirable to attempt to remedy the breach to the Secured Party's satisfaction. The Secured Party is not obliged to do so. Any liabilities or expenses incurred by the Secured Party in attempting to remedy any such breach must be reimbursed by the Grantor on demand and, pending reimbursement, will be part of the Secured Money.
- (b) Clause 6.3(a) does not limit any other right the Secured Party has under this document or at law.

6.4 Powers on enforcement

If this document has become enforceable, the Secured Party or any of its Authorised Representatives, without notice to the Grantor, may:

- (a) exercise any of the powers that might be exercised by a Receiver even if a Receiver has not been appointed; and
- (b) complete any Transfer, other transfer or instrument of any nature executed by or on behalf of the Grantor in blank and deposited with the Secured Party as Collateral Security in favour of, or take any other action required to transfer any Secured Property to, the Secured Party or any appointee of the Secured Party or any other person.

POWER OF ATTORNEY

7.1 Appointment of Attorneys

The Grantor irrevocably appoints the Secured Party and each Authorised Representative of the Secured Party and as an independent appointment appoints any Receiver, severally its attorney, at the Grantor's cost, to:

- (a) (all acts necessary) do anything necessary or desirable in the opinion of the Secured Party or the Attorney to:
 - (i) complete this document;
 - (ii) give full effect to this document;
 - (iii) better secure the Secured Property to the Secured Party in a manner consistent with this document; or
 - (iv) assist in the execution or exercise of any power under this document,

including give directions to the Nominee or any Controlling Participant in respect of CHESS Securities in the Secured Property, and execute any Transfer (including a Transfer in blank) or other document;

- (b) do all or any of the following while an Event of Default in respect of which the Grantor is the Defaulting Party is continuing:
 - (i) (recover Secured Property) demand, sue for, recover and give discharge for the Secured Property;
 - (ii) (commence actions) commence, carry on , enforce, settle, arrange and compromise any proceedings to obtain or enforce the payment or delivery of the Secured Property;
 - (iii) (compound debts) compound, settle or compromise any debt of the Grantor in connection with the Secured Property;
 - (iv) (execute deeds) execute any agreement including any deed of assignment, composition or release in connection with the Secured Property;
 - (v) (exercise Marketable Security owner's rights) exercise all or any powers, rights, discretions and remedies available to any owner or holder (whether beneficial or otherwise) of any Marketable Securities included in the Secured Property (including any right to give directions to the Nominee or any Controlling Participant or to redeem any Marketable Security and rights available to an owner of Marketable Securities under the Corporations Act or any other statute, or under the rules of any stock exchange); and
 - (vi) (**general**) do anything else that the Grantor must or may do, or that the Secured Party may do, under this document or by law.

7.2 General

- (a) Each Attorney may appoint and remove substitutes, and may delegate its powers (including this power of delegation) and revoke any delegation.
- (b) An Attorney may do anything contemplated by this clause even if the Attorney is affected by an actual or potential conflict of interest or duty, or might benefit from doing it.

- (c) An Attorney may do anything contemplated by this clause in its name, in the name of the Grantor or in the name of both of them.
- (d) The Grantor must ratify anything done by an Attorney under this clause.
- (e) The Grantor gives the power of attorney in this clause:
 - (i) to secure performance by the Grantor of its obligations to the Secured Party under this document and any property interest of the Secured Party under this document; and
 - (ii) for valuable consideration, receipt of which is acknowledged by the Grantor.

8. **ENFORCEMENT**

8.1 Circumstances when this document may be enforced

The Secured Money will immediately become payable at the Secured Party's option (despite any delay or previous waiver of the right to exercise that option) without the need for any demand or notice under this document or under another Equity Derivative Financing Document, and this document will immediately become enforceable (whether or not the Secured Money has become payable in this manner) if an Event of Default occurs.

8.2 Enforcement despite earlier payment

This document may be enforced:

- (a) even if the Secured Party accepts a payment of interest or other amount after the occurrence of any Event of Default; and
- (b) without the need for any notice to, or of any consent or agreement of, the Grantor or any other person.

8.3 Exclusion of PPSA provisions

To the extent the law permits:

- (a) for the purposes of sections 115(1) and 115(7) of the PPSA:
 - (i) the Secured Party need not comply with sections 95, 118, 121(4), 125, 130, 132(3)(d) or 132(4); and
 - (ii) sections 142 and 143 are excluded;
- (b) for the purposes of section 115(7) of the PPSA, the Secured Party need not comply with sections 132 and 137(3); and
- (c) if the PPSA is amended after the date of this document to permit the Grantor and the Secured Party to agree to not comply with or to exclude other provisions of the PPSA, the Secured Party may notify the Grantor that any of these provisions is excluded, or that the Secured Party need not comply with any of these provisions, as notified to the Grantor by the Secured Party.

9. **DEFAULT INTEREST**

9.1 Grantor must pay interest

(a) The Grantor must pay interest on each amount that is not paid when due (unless the Grantor is already required to pay interest on the unpaid amount by the terms

of any other Equity Derivative Financing Document), from (and including) the day on which it falls due to (but excluding) the day on which it is paid in full, at the rate calculated in accordance with clause 9.1(b). This interest must be paid on demand.

- (b) Interest on an unpaid amount accrues each day at a rate equal to the rate specified in the Bridge Facility Agreement for the payment by the Grantor of interest on unpaid amounts, and is capitalised (if not paid) on the last day of each Default Interest Period (as defined in the Bridge Facility Agreement).
- (c) This subclause does not affect the Grantor's obligation to pay each amount under this document when it is due.

9.2 Interest after judgment

If a liability of the Grantor becomes merged in a judgment or order, the Grantor, as an independent obligation, must pay interest on the amount of that liability, from (and including) the date of the judgment or order until it is paid in full, at the higher of the rate that applies under the judgment or order and the rate calculated in accordance with clause 9.1.

9.3 Accrual and calculation of interest

Interest under this clause:

- (a) accrues daily; and
- (b) is calculated on the basis of the actual number of days on which interest has accrued and of a 365 day year.

10. APPOINTMENT OF RECEIVER

10.1 Appointment

If this document has become enforceable (whether or not the Secured Party has entered into possession of all or any of the Secured Property) the Secured Party or any Authorised Representative of the Secured Party may at any time:

- appoint any person or any two or more persons jointly and severally to be a receiver or receiver and manager (or an additional receiver or receiver and manager) of Secured Property;
- (b) remove the Receiver and in case of the removal, retirement or death of any Receiver appoint another as a replacement; and
- (c) fix the remuneration of the Receiver.

Subject to clause 10.2, every Receiver appointed under this subclause will be the Grantor's agent and the Grantor alone will be responsible for the Receiver's acts and defaults and remuneration.

10.2 Receiver other than as Grantor's agent

- (a) The Secured Party by notice to the Grantor and the Receiver may require the Receiver to act as the Secured Party's agent.
- (b) The power to appoint a Receiver under this clause may be exercised even though:

- (i) at the time when this document becomes enforceable or when an appointment is made, an order may have been made or a resolution may have been passed to wind up the Grantor; or
- (ii) a Receiver appointed in the circumstances specified in the preceding paragraph may not, or may not in some respects, act as the Grantor's agent.

10.3 Powers of Receiver

The Receiver will have full power to do all or any of the following:

- (a) (possession and control) take possession of, or take control of, collect and get any Transfers or other documents held or entitled to be held by the Secured Party under the Security Interest granted in clause 2.1 and the Secured Property and for that purpose to take proceedings (in the name of the Grantor or otherwise);
- (b) (**give up possession**) give up possession of any Transfers or other documents held or entitled to be held by the Secured Party under the Security Interest granted in clause 2.1 and the Secured Property;
- (c) (exercise Secured Party's rights)
 - exercise all or any of the Secured Party's powers, rights, discretions and remedies under this document; and
 - (ii) comply with the directions given by the Secured Party;
- (d) (exercise Marketable Security owner's rights) exercise all or any powers, rights, discretions and remedies available to any owner or holder (whether beneficial or otherwise) of any Marketable Securities included in the Secured Property (including any right to give directions to the Nominee or any Controlling Participant or to redeem any Marketable Security and rights available to an owner of Marketable Securities under the Corporations Act or any other statute, or under the rules of any stock exchange);
- (e) (**registration**) do everything necessary to obtain registration of the Secured Property in the name of the Secured Party or its nominee and to enable the Secured Party or its nominee to receive New Rights;
- (f) (New Rights) receive all New Rights;
- (g) (settle disputes)
 - (i) settle, arrange and compromise any accounts, claims, questions or disputes that may arise in connection with the Secured Property or in any way relating to this document; and
 - execute releases or other discharges in relation to the settlement, arrangement, or compromise;
- (h) (sell) sell (whether or not the Receiver has taken possession), exchange or otherwise dispose of (absolutely or conditionally) the Secured Property (or agree to do so):
 - (i) with or without any other property;
 - (ii) by public auction, private sale or tender for cash or on credit;

- (iii) whether or not the reserve price for a sale by auction or tender is disclosed;
- (iv) in one lot or in parcels;
- (v) with or without special conditions, (such as conditions as to title or time or method of payment of purchase money) including by allowing the purchase money to remain:
 - (A) outstanding on any security over the property sold or over any other property; or
 - (B) owing without any security; and
- (vi) on other terms the Receiver considers desirable,

without being responsible for any loss;

- (i) (**transfer on sale**) execute Transfers, other transfers and assignments of Secured Property (including in the name of the Grantor), and do everything to complete any sale under clause 10.3(h) that the Receiver thinks necessary;
- (j) (employees and agents) engage employees, agents, consultants, lawyers, advisers and contractors for any of the purposes of this clause on terms that the Receiver thinks appropriate;
- (k) (**give receipts**) give receipts for all money and other property that may come into the hands of the Receiver in exercise of any power given by this document;
- (enforce contracts) carry out and enforce or otherwise obtain the benefit of all contracts:
 - (i) entered into or held by the Grantor in connection with the Secured Property;
 - (ii) entered into in exercise of the powers given by this document;
- (m) (make debtors bankrupt) wind up the issuer and do everything in connection with any winding up that the Receiver thinks desirable to recover or protect Secured Property;
- (n) (perform undertakings) do everything necessary to perform any undertaking of the Grantor in this document;
- (o) (receive money) receive all money or other property payable or deliverable to the Grantor from the Secured Property;
- (p) (desirable or incidental matters)
 - (i) do or cause to be done everything that the Receiver thinks desirable in the interests of the Secured Party; and
 - (ii) do anything incidental to the exercise of any other power;
- (q) (take legal proceedings) take proceedings (including in the name of the Grantor) in connection with any of the above; and
- (r) (**delegate**) with the Secured Party's consent delegate any of the powers given to the Receiver by this clause to any person.

11. PROTECTION OF SECURED PARTY AND APPOINTEES

11.1 Protection of Secured Party and Receiver

- (a) The Secured Party is not obliged to, but may, do the following:
 - (i) notify any debtor or member of the Grantor or any other person of this document; or
 - (ii) enforce payment of any money payable to the Grantor, or take any step or proceeding for any similar purpose.
- (b) None of the Secured Party, any of its Authorised Representatives or agents, any Attorney or any Receiver is liable for any omission or delay in exercising any power, right, discretion or remedy under this document or for any involuntary loss or irregularity that may occur in relation to the exercise or non-exercise of any of them except to the extent that it is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

11.2 Conflict of interests

The Secured Party, an Authorised Representative or agent of the Secured Party, an Attorney, Receiver or other person appointed by the Secured Party under this document may exercise or agree to exercise a power given by this document or by law even though that person may have a conflict of interests in exercising the power.

11.3 Liability for loss

- (a) None of the Secured Party, an Authorised Representative or agent of the Secured Party, an Attorney, a Receiver or any other person appointed by the Secured Party under this document is liable for any loss that the Grantor suffers as a direct or indirect result of:
 - (i) the exercise or attempted exercise of, or failure to exercise, any of its rights contained in this document; and
 - (ii) any release or dealing with any other Guarantee or Security Interest (including any prejudice to or loss of the Grantor's rights of subrogation),

except to the extent that such loss is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

- (b) If the Secured Party, any agent of the Secured Party or a Receiver enters into possession of Secured Property, none of the Secured Party, any of its Authorised Representatives or agents, any Attorney or any Receiver is liable:
 - (i) to account as secured party in possession or for anything except actual receipts; or
 - (ii) for any loss on realisation or for any default or omission for which a secured party in possession might be liable, except to the extent that it is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

12. PROTECTION OF THIRD PARTIES

12.1 Dealings under this document

A purchaser or other party to a disposal or dealing in attempted exercise of a power contained in this document is not:

- (a) bound to enquire whether an Event of Default has occurred, whether this document has become enforceable, whether a Receiver has been properly appointed or about the propriety or regularity of a sale, disposal or dealing; or
- (b) affected by notice that a sale, disposal or dealing is unnecessary or improper.

Despite any irregularity or impropriety in a sale, disposal or dealing, it is to be treated, for the protection of the purchaser or other party to the disposal or dealing, as being authorised by this document and valid.

12.2 Receipts

A receipt that the Secured Party, one of its Authorised Representatives or agents or a Receiver gives for any money payable to or receivable by the Secured Party or the Receiver because of this document will:

- (a) relieve the person paying or handing over money or other property from all liability:
 - (i) for the application (or any loss or misapplication) of the money or other property;
 - (ii) to enquire whether the Secured Money has become payable; and
 - (iii) (where appropriate) as to the propriety or regularity of the appointment of the Receiver; and
- (b) discharge the person paying that money from its liability to pay that money.

13. APPLICATION OF MONEY

13.1 **Order**

- (a) Money that the Secured Party or a Receiver receives under or because of this document is to be applied, after satisfaction of any claims that the Secured Party or the Receiver is aware is a claim that ranks in priority to the Security Interest granted in clause 2.1, in the following order or in any other order that the Secured Party or the Receiver may determine in its absolute discretion, subject to any Equity Derivative Financing Documents or applicable law to the contrary (for example, section 140 of the PPSA where relevant):
 - (i) (expenses) first in payment of all expenses that the Secured Party or a Receiver incurs in or incidental to the exercise or attempted exercise of a power or otherwise in relation to any Equity Derivative Financing Document;
 - (ii) (outgoings) then in payment of any other outgoings that the Receiver or the Secured Party thinks it appropriate to pay;
 - (iii) (**Receiver**) then in payment to the Receiver of any remuneration (whether by way of commission or otherwise);

- (iv) (indemnities) then in payment to the Secured Party or a Receiver of any amount necessary to give effect to any indemnity contained in this document; and
- (v) (Secured Money) then in payment to the Secured Party of the Secured Money.
- (b) Any surplus will belong to the Grantor or other persons entitled to it. The Secured Party or the Receiver may pay the surplus to the credit of a bank account in the name of the Grantor or other person entitled to it or into court and will then be under no further liability in relation to it. The surplus will not accrue interest.
- (c) When the Secured Party or a Receiver receives money under or because of this document, and applies it in payment to the Secured Party of the Secured Money, the Secured Party or the Receiver (as the case may be) may apply different parts of the money received to different parts of the Secured Money, regardless of any appropriation by the Grantor, as the Secured Party or Receiver chooses.

13.2 Only actual receipts credited

In applying any money towards the Secured Money, the Grantor's account will be credited only with the amount of the money that the Secured Party actually receives for that purpose. The credit will date from the time of receipt.

13.3 Compensation

If any compensation becomes payable for the Secured Property, the Secured Party may:

- (a) apply the sum received on account of any compensation, at the Secured Party's option, in or towards repayment of the Secured Money;
- (b) make, enforce, settle or compromise any claims relating to compensation; and
- (c) execute any necessary assurances and releases in the names of the Grantor and the Secured Party.

If any compensation comes into the hands of the Grantor before a final irrevocable discharge of this document, the Grantor must immediately pay it to the Secured Party.

13.4 Certificates and disputes

- (a) The Secured Party may rely on a certificate issued by any person who claims to be entitled to any money received from the exercise of any right in relation to the Secured Property which states that the Grantor owes it a certain amount of money, without making any further enquiry.
- (b) If there is any dispute between any persons (other than the Secured Party) regarding an entitlement to receive any money received from the exercise of any right in relation to the Secured Property, the Secured Party may pay that money into court, and after doing so does not have any further obligation in respect of that money.

13.5 No interest

The Secured Party is not obliged to pay interest to any person on any money received from the exercise of any right in relation to the Secured Property.

13.6 Payment into bank account

The Secured Party or the Receiver may pay any money to the credit of a bank account in the name of a person to whom it is obliged to pay any money received from the exercise of any right in relation to the Secured Property, and having done so is under no further liability in respect of that money.

13.7 Amounts contingently due

- (a) If any part of the Secured Money is contingently owing to the Secured Party when money is being applied under clause 13.1 the Secured Party or Receiver may:
 - retain an amount equal to the amount contingently owing, or any part of it;and
 - (ii) put that amount in an interest-bearing account, payable at call.
- (b) If the amount which is contingently owing:
 - (i) becomes payable; or
 - (ii) ceases to be contingently owing,

the Secured Party or Receiver must apply the amount retained (and any interest earned on it) in accordance with clause 13.1.

14. CONTINUING SECURITY

14.1 Continuity

The Security Interest granted in clause 2.1:

- is a continuing security, and remains in full force until a final irrevocable discharge
 of that Security Interest is given to the Grantor under clause 17 despite any
 transaction or other thing (including a settlement of account or intervening
 payment); and
- (b) will apply to the present and future balance of the Secured Money.

14.2 Limitations on Grantor's rights

Until the Secured Money has been irrevocably paid and discharged in full, the Grantor may not:

- (a) share in any Guarantee, Security Interest or money received or receivable by the Secured Party in relation to the Secured Money or stand in the place of the Secured Party in relation to any Guarantee, Security Interest or right to receive money;
- (b) in reduction of its liability under this document, raise a defence, set off or counterclaim against the Secured Party or claim a set off or make a counterclaim against the Secured Party; or
- (c) claim to be entitled by way of contribution, indemnity, subrogation, marshalling or otherwise to the benefit of any document or agreement to which the Secured Party is a party.

14.3 No marshalling

The Secured Party is not under any obligation to marshal or appropriate in favour of the Grantor or to exercise, apply, perfect or recover any Security Interest that the Secured Party holds at any time or any funds or property that the Secured Party may be entitled to receive or have a claim on.

14.4 Effect of Insolvency Event

- (a) If an Insolvency Event has occurred in relation to an Obligor, any amount paid by that Obligor (relevant payment) will only be applied against any Secured Money if:
 - the Secured Party forms the opinion in good faith (which will be conclusively binding on the Grantor) that it will not be required to pay the relevant payment to any person under any law relating to bankruptcy, winding up or the protection of creditors; or
 - (ii) a final judgment is given by a court of competent jurisdiction in favour of the Secured Party that it is not required to pay the relevant payment to any person under any law relating to bankruptcy, winding up or the protection of creditors.
- (b) If an amount is applied against any Secured Money and the Secured Party pays or determines that it is obliged to pay the relevant amount to any person under any law relating to bankruptcy, winding up or the protection of creditors:
 - (i) the Secured Party's rights are to be reinstated and will be the same in relation to that amount as if the application, or the payment or transaction giving rise to it, had not been made; and
 - (ii) the Grantor must immediately do anything (including the signing of documents) required by the Secured Party to restore to the Secured Party any Guarantee or Security Interest to which it was entitled immediately before that application or the payment or transaction giving rise to it.
- (c) Any discharge or release between the Secured Party and the Grantor is subject to reinstatement of the Secured Party's rights under this subclause.

15. **RECOVERY OF GST**

- (a) Unless otherwise indicated all amounts referred to in this document are exclusive of GST.
- (b) If the Secured Party makes a taxable supply under or in connection with this document for consideration that is exclusive of GST, the Grantor must:
 - pay to the Secured Party an amount equal to any GST for which the Secured Party is liable in relation to that supply; and
 - (ii) make that payment as and when the consideration or part of it must be paid or provided.
- (c) If requested by the Grantor, the Secured Party must issue a tax invoice for a taxable supply to the person to whom it made the supply.
- (d) The Grantor's obligation to reimburse the Secured Party for an amount paid or payable to a third party (including an obligation to pay the Secured Party's or another party's legal costs) includes GST on the amount paid or payable to the

third party except to the extent that the Secured Party is entitled to an input tax credit for that GST. Unless notified otherwise by the Secured Party, the Grantor must assume that the Secured Party is not entitled to any input tax credit for that GST.

16. **INDEMNITIES**

16.1 General indemnities

The Grantor must indemnify the Secured Party against, and must pay the Secured Party on demand the amount of, all losses (including loss of profit), liabilities, costs, expenses and Taxes (other than Excluded Taxes) incurred in connection with each of the following:

- (a) any Secured Money being paid or becoming due for payment other than on its due date, or any amount required to be paid under any Equity Derivative Financing Document not being paid on its due date, including losses (including loss of profit), liabilities, costs, expenses and Taxes (other than Excluded Taxes) incurred because of:
 - (i) the cancellation, termination, unwinding or alteration of any swap or other arrangement made by the Secured Party to fund the Secured Money or other amount; or
 - (ii) any liquidation or re-employment of deposits or other funds acquired by the Secured Party to fund the Secured Money or other amount;
- (b) the preparation, negotiation, execution, stamping and registration of the Equity Derivative Financing Documents and the satisfaction of any conditions precedent;
- (c) the transactions that the Equity Derivative Financing Documents contemplate;
- (d) any amendment to, or any consent, approval, waiver, release or discharge of or under, an Equity Derivative Financing Document;
- (e) any Event of Default or Potential Event of Default;
- (f) the administration, and any actual or attempted preservation or enforcement, of any rights under the Equity Derivative Financing Documents;
- (g) an enquiry by a Government Agency involving the Grantor; and
- (h) any action taken by the Secured Party under or in relation to the PPSA, including any registration, or any response to an amendment demand or a request under section 275 of the PPSA,

including legal expenses on a full indemnity basis (provided that the Borrower shall only be liable for reasonable legal expenses incurred in connection with paragraph (b)), the Secured Party's internal administration and legal costs at the rate and on the basis determined by the Secured Party and expenses incurred by the Secured Party in engaging investigative accountants and financial consultants (but only where such engagement is made after the occurrence of an Event of Default or a Potential Event of Default).

16.2 Indemnity for exercise of rights or proceedings

To the extent permitted by law, the Grantor must indemnify each of the Secured Party, each Authorised Representative and agent of the Secured Party, each Receiver and Attorney of the Grantor and any other person appointed under this document or the Corporations Act by or on behalf of the Secured Party as secured party under this

document against, and must pay each of them on demand the amount of all losses, liabilities, costs, expenses and Taxes (other than Excluded Taxes) that they each incur:

- (a) (directly or indirectly) in the exercise or attempted exercise of any of the powers, rights, discretions or remedies (express or implied) vested in them under this document or the Corporations Act; and
- (b) in connection with all proceedings, expenses, claims and demands in relation to anything done or omitted in any way relating to Secured Property,

including legal expenses on a full indemnity basis and expenses incurred in engaging investigative accountants and financial consultants (but only where such engagement is made after the occurrence of an Event of Default or a Potential Event of Default)except to the extent that any such losses, liabilities, costs, expenses and Taxes are a direct and foreseeable result of the fraud, gross negligence or wilful misconduct of that person.

16.3 Currency indemnity

If, for any reason (including as a result of a judgment or order), an amount payable by the Grantor under or in respect of this document (**Relevant Amount**) is received by the Secured Party in a currency (**Payment Currency**) that is not the currency in which the amount is expressed to be payable under this document (**Required Currency**) then the Grantor, as an independent obligation, must indemnify the Secured Party against, and must pay the Secured Party on demand the amount of, any shortfall between:

- (a) the amount of Required Currency which the Secured Party receives on converting the amount it received in the Payment Currency into an amount in the Required Currency in accordance with its usual practice; and
- (b) the Relevant Amount in the Required Currency.

16.4 Recovery from Secured Property

A person who is entitled to be indemnified for a loss, liability, expense or Tax under clauses 16.1, 16.2 or 16.3 may recover the amount to be indemnified direct from the Secured Property.

17. **DISCHARGE**

The Secured Party must at the request and cost of the Grantor reconvey, surrender or release any remaining Secured Property (as appropriate) to the Grantor and the Secured Property will then be discharged from this document:

- (a) when the Secured Party is satisfied that:
 - (i) all the Secured Money has been irrevocably paid and discharged in full; and
 - (ii) no amount remains contingently payable or may become payable on the security of this document (including under an indemnity); and
- (b) on payment or retention of all expenses incurred by or payable to the Secured Party, its Authorised Representatives or any Receiver or Attorney.

Any discharge is subject to clause 14.4.

18. **CONFIDENTIALITY**

18.1 General confidentiality

The parties to this document agree that the Secured Party must not disclose any information provided by the Grantor under the Equity Derivative Financing Documents to any person who is not a party, and the Grantor must not disclose any information concerning the contents of or the transactions contemplated by any Equity Derivative Financing Document, including in each case information of the kind mentioned in section 275(1) of the PPSA, to any person who is not a party, except to the extent that:

- (a) (**permitted by documents**) the disclosure is expressly permitted by an Equity Derivative Financing Document or required by law to ensure the enforceability or priority of any Equity Derivative Financing Document;
- (b) (consent of other party) the other party consents to the disclosure;
- (c) (public domain) the information is already in the public domain, unless it entered the public domain because of a breach of confidentiality by the disclosing party;
- (d) (employees and advisers) the disclosure is made on a confidential basis to the officers, employees, agents, lawyers, auditors or other professional advisers of the party or its holding company. For the avoidance of doubt, it is not mandatory for a receiving party to enter into any written confidentiality undertaking in respect of any disclosure made under this paragraph;
- (e) (**comply with laws**) the disclosure is necessary to comply with any applicable law, or an order of a court or tribunal;
- (f) (**comply with directives**) the disclosure is necessary to comply with a directive or request of any Government Agency (whether or not having the force of law);
- (g) (**obtain Authorisations**) the disclosure is necessary or desirable to obtain an Authorisation from any Government Agency;
- (h) (discovery and litigation) the disclosure is necessary or desirable in relation to any discovery of documents, or any proceedings before a court, tribunal or other Government Agency; or
- (i) (**Secured Party's purposes**) it is in the Secured Party's interest to disclose the information, including:
 - (i) to persons considering purchasing the Secured Party or who are involved in valuing the Secured Party for ratings purposes; or
 - (ii) to persons considering taking an assignment of, interest in or exposure to the Facility or the Equity Derivative Financing Documents, including under securitisation or sub-participation arrangements.

Any disclosure made under this clause 18.1(i) must be made on the basis that the receiving party must keep the information confidential as required by this clause.

18.2 No request

The Grantor agrees not to exercise its rights to make any request of the Secured Party under section 275 of the PPSA, to authorise the disclosure of any information under that section or to waive any duty of confidence that would otherwise permit non-disclosure under that section (but this does not limit the Grantor's rights to request information other than under section 275).

19. **NOTICES**

19.1 How to give a notice

A notice, consent or other communication under this document is only effective if it is:

- (a) in writing, signed by or on behalf of the person giving it;
- (b) addressed to the person to whom it is to be given; and
- (c) given in one of the following ways:
 - (i) sent by prepaid mail (by airmail, if the addressee is overseas) or delivered to that person's address;
 - (ii) sent by fax to that person's fax number and the machine from which it is sent produces a report that states that it was sent in full without error;
 - (iii) given personally; or
 - (iv) given in any other manner permitted by law.

19.2 When a notice is given

Subject to clause 19.3, a notice, consent or other communication that complies with this clause is conclusively regarded as given and received:

- (a) if it is sent by fax or delivered, if received:
 - (i) by 5.00 pm (local time in the place of receipt) on a Business Day on that day; or
 - (ii) after 5.00 pm (local time in the place of receipt) on a Business Day, or on a day that is not a Business Day on the next Business Day;
- (b) if it is sent by mail, when it would be delivered in the ordinary course of post, but in any event:
 - (i) not later than three Business Days after posting within Australia; or
 - (ii) not later than seven Business Days after posting to or from a place outside Australia;
- (c) if given personally, when actually received by that person; and
- (d) if it is given in any other manner permitted by law, when actually received by that person, unless a later time of receipt is specified in it.

19.3 Notices to the Secured Party

Any notice, consent or other communication to be made or delivered to the Secured Party will be effective:

- (a) only when actually received by it; and
- (b) only if it is expressly marked for the attention of the department or officer specified in the following clause (or any substitute department or officer as it specifies for this purpose).

19.4 Address for notices

A person's address and fax number are those set out below, or as the person notifies the sender:

Secured Party

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61-2-9324-2880

Email: <u>elaine.chow@ubs.com</u> / <u>kevin.bayett@ubs.com</u> / <u>sh-aus-eq-</u>

support@ubs.com / michael.rollin@ubs.com / sh-syd-rmp-op@ubs.com

Attention: Elaine Chow / Kevin Bayett / Michael Rollin

Grantor

Address: The Independent State of Papua New Guinea

Department of Treasury

P O Box 542 Waigani 131 N.C.D.

Papua New Guinea

Telephone: +675 3235600

Fax number: +675 3128804

Email: dairi_vele@treasury.gov.pg

Attention: Mr Dairi Vele, Secretary for Department of Treasury

20. AMENDMENT AND ASSIGNMENT

20.1 Amendment

This document can only be amended or replaced by another document executed by the parties.

20.2 **Assignment**

- (a) The Grantor may only assign, encumber, declare a trust over or otherwise deal with its rights under any Equity Derivative Financing Document with the written consent of the Secured Party.
- (b) The Secured Party may assign, encumber, declare a trust over or otherwise deal with its rights under this document without the consent of the Grantor, and may disclose to any potential holder of the right, or an interest in the right, any information relating to this document or any party to it.

21. **GENERAL**

21.1 Governing law

- (a) This document is governed by the laws of the State of New South Wales.
- (b) Each party submits to the non-exclusive jurisdiction of the courts of the State of New South Wales, and any court that may hear appeals from any of those courts, for any proceedings in connection with any Equity Derivative Financing Document.
- (c) The Grantor irrevocably waives:
 - (i) any objection to the venue of any proceedings on the ground that they have been brought in an inconvenient forum; and
 - (ii) any immunity from set off, suits, proceedings and execution to which it or any of its property may now or in the future be entitled under any applicable
- (d) The Grantor confirms the appointment under the "Process Agent" provisions of the Equity Derivative Financing Agreement of Norose Notices Australia Pty Ltd ACN 158 029 586 of Level 16, Grosvenor Place, 225 George Street, Sydney, NSW 2000 as its agent to receive service of process for any proceedings in connection with any Equity Derivative Financing Document. The Grantor undertakes to maintain this appointment until all Secured Money is repaid, and agrees that any process served on that person is taken to be served on it.

21.2 Liability for own expenses

The Grantor is liable for its own costs and expenses in complying with this document, including where it does so at the Secured Party's request or for the Secured Party's benefit.

21.3 Giving effect to this document

- (a) The Grantor must do anything (including executing any Transfer and other transfer in blank, or any other document, and perfecting and protecting any Security Interest intended to be created by or pursuant to this document), and must ensure that its employees and agents do anything, that the Secured Party may reasonably require to:
 - (i) give full effect to this document; or
 - (ii) more fully secure the rights, remedies and powers of the Secured Party under this document or to enable the Secured Party to exercise those rights, remedies and powers,

including giving directions to the Nominee or any Controlling Participant in respect of any CHESS Securities in the Secured Property.

(b) The Secured Party may, at the Grantor's cost, do anything which the Grantor should have done under this document if the Grantor does not do so promptly or, if in the Secured Party's opinion, the Grantor does not do so properly.

21.4 Authority to register and waiver of right to receive verification statements

The Grantor acknowledges that the Secured Party may, at the Grantor's cost, register one or more financing statements in relation to its Security Interest. If permitted by the PPSA, the Grantor waives its right under section 157 of the PPSA to receive notice of any

verification statement relating to the registration of any such financing statement or any related financing change statement.

21.5 Variation of rights

The exercise of a right partially or on one occasion does not prevent any further exercise of that right in accordance with the terms of this document. Neither a forbearance to exercise a right nor a delay in the exercise of a right operates as an election between rights or a variation of the terms of this document.

21.6 **Operation of this document**

- (a) Subject to clause 21.6(b), the Equity Derivative Financing Documents contain the entire agreement between the parties about their subject matter. Any previous understanding, agreement, representation or warranty relating to that subject matter is replaced by the Equity Derivative Financing Documents and has no further effect.
- (b) Any right that the Secured Party may have under the Equity Derivative Financing Documents is in addition to, and does not replace or limit, any other right that the Secured Party may have.
- (c) Any provision of this document which is unenforceable or partly unenforceable is, where possible, to be severed to the extent necessary to make this document enforceable, unless this would materially change the intended effect of this document.

21.7 Operation of indemnities

- (a) Each indemnity in this document survives the expiry or termination of this document.
- (b) The Secured Party may recover a payment under an indemnity in this document before it makes the payment in respect of which the indemnity is given.
- (c) If a provision of this document is expressed to:
 - (i) indemnify;
 - (ii) exclude or limit any liability of; or
 - (iii) otherwise benefit,

a person who is not a party to this document, the Grantor agrees that the Secured Party holds the benefit of that indemnity, exclusion, limitation or other benefit for that person and may enforce this document on their behalf and for their benefit.

21.8 Consents

Where this document contemplates that the Secured Party may agree or consent to something (however it is described), the Secured Party may:

- (a) agree or consent, or not agree or consent, in its absolute discretion; and
- (b) agree or consent subject to conditions,

unless this document expressly contemplates otherwise.

21.9 Statements by Secured Party

A statement by an Authorised Representative of the Secured Party on any matter relating to any Equity Derivative Financing Document (including any amount owing by the Grantor) is conclusive unless clearly wrong on its face.

21.10 Set-off

If an Event of Default occurs and is continuing, the Secured Party, without notice to the Grantor, may combine any account that the Grantor holds at any branch or office (in Australia or elsewhere) of the Secured Party with, or set off any amount in any currency that is or may become owing in any currency by the Secured Party to the Grantor against, any amount owing by the Grantor to the Secured Party under the Equity Derivative Financing Documents. For this purpose the Secured Party may:

- (a) change the terms (including the repayment date) of any account or other payment obligation between the parties;
- (b) convert amounts into different currencies in accordance with the Secured Party's usual practice; and
- (c) do anything (including execute any document) in the name of the Grantor that the Secured Party considers necessary or desirable.

This subclause overrides any other document or agreement to the contrary.

21.11 No merger

Nothing in this document merges with any other Security Interest, or any Guarantee, judgment or other right or remedy, that the Secured Party may hold at any time.

21.12 Exclusion of contrary legislation

Any legislation that affects an obligation of the Grantor in a manner that is adverse to the interests of the Secured Party, or adversely affects the exercise by the Secured Party of a right or remedy, under or relating to this document is excluded to the full extent permitted by law.

21.13 Counterparts

This document may be executed in counterparts. Delivery of a counterpart of this document by email attachment or fax constitutes an effective mode of delivery.

21.14 Execution by fewer than all parties

This document binds the Grantor even if the Secured Party does not execute or only subsequently executes this document.

EXECUTED as a deed.

Each person who executes this document on behalf of a party under a power of attorney declares that he or she is not aware of any fact or circumstance that might affect his or her authority to do so under that power of attorney.

Signed, Sealed and Delivered for and on behalf of the **INDEPENDENT STATE OF PAPUA NEW GUINEA** by Sir Michael Oglo, G.C.M.G., K.St.J., Governor General, acting with and in accordance with the advice of the National Executive Council:

CACHO

in the presence of:

Name: CARL OKUK

Title: JAMER

Signed, Sealed and Delivered for **UBS AG, Australia Branch** by its duly authorised signatories in the presence of:

Witness Signature KEVIN LU

Print Name

Authorised Signatory Signature

Jason McDonnell

Print Name

Authorised Signatory Signature

Print Name

34



International Swaps and Derivatives Association, Inc.

CREDIT SUPPORT ANNEX

to the Schedule to the

ISDA Master Agreement

dated as of	
be	etween
	and
("Party A")	("Party B")

This Annex supplements, forms part of, and is subject to, the ISDA Master Agreement referred to above and is part of its Schedule. For the purposes of this Agreement, including, without limitation, Sections 1(c), 2(a), 5 and 6, the credit support arrangements set out in this Annex constitute a Transaction (for which this Annex constitutes the Confirmation).

Paragraph 1. Interpretation

Capitalised terms not otherwise defined in this Annex or elsewhere in this Agreement have the meanings specified pursuant to Paragraph 10, and all references in this Annex to Paragraphs are to Paragraphs of this Annex. In the event of any inconsistency between this Annex and the other provisions of this Schedule, this Annex will prevail, and in the event of any inconsistency between Paragraph 11 and the other

This document is not intended to create a charge or other security interest over the assets transferred under its terms. Persons intending to establish a collateral arrangement based on the creation of a charge or other security interest should consider using the ISDA Credit Support Deed (English law) or the ISDA Credit Support Annex (New York law), as appropriate.

This Credit Support Annex has been prepared for use with ISDA Master Agreements subject to English law. Users should consult their legal advisers as to the proper use and effect of this form and the arrangements it contemplates. In particular, users should consult their legal advisers if they wish to have the Credit Support Annex made subject to a governing law other than English law or to have the Credit Support Annex subject to a different governing law than that governing the rest of the ISDA Master Agreement (e.g., English law for the Credit Support Annex and New York law for the rest of the ISDA Master Agreement).

provisions of this Annex, Paragraph 11 will prevail. For the avoidance of doubt, references to "transfer" in this Annex mean, in relation to cash, payment and, in relation to other assets, delivery.

Paragraph 2. Credit Support Obligations

- (a) **Delivery Amount.** Subject to Paragraphs 3 and 4, upon a demand made by the Transferee on or promptly following a Valuation Date, if the Delivery Amount for that Valuation Date equals or exceeds the Transferor's Minimum Transfer Amount, then the Transferor will transfer to the Transferee Eligible Credit Support having a Value as of the date of transfer at least equal to the applicable Delivery Amount (rounded pursuant to Paragraph 11(b)(iii)(D)). Unless otherwise specified in Paragraph 11(b), the "Delivery Amount" applicable to the Transferor for any Valuation Date will equal the amount by which:
 - (i) the Credit Support Amount

exceeds

- (ii) the Value as of that Valuation Date of the Transferor's Credit Support Balance (adjusted to include any prior Delivery Amount and to exclude any prior Return Amount, the transfer of which, in either case, has not yet been completed and for which the relevant Settlement Day falls on or after such Valuation Date).
- (b) **Return Amount.** Subject to Paragraphs 3 and 4, upon a demand made by the Transferor on or promptly following a Valuation Date, if the Return Amount for that Valuation Date equals or exceeds the Transferee's Minimum Transfer Amount, then the Transferee will transfer to the Transferor Equivalent Credit Support specified by the Transferor in that demand having a Value as of the date of transfer as close as practicable to the applicable Return Amount (rounded pursuant to Paragraph 11(b)(iii)(D)) and the Credit Support Balance will, upon such transfer, be reduced accordingly. Unless otherwise specified in Paragraph 11(b), the "Return Amount" applicable to the Transferee for any Valuation Date will equal the amount by which:
 - (i) the Value as of that Valuation Date of the Transferor's Credit Support Balance (adjusted to include any prior Delivery Amount and to exclude any prior Return Amount, the transfer of which, in either case, has not yet been completed and for which the relevant Settlement Day falls on or after such Valuation Date)

exceeds

(ii) the Credit Support Amount.

Paragraph 3. Transfers, Calculations and Exchanges

- (a) *Transfers*. All transfers under this Annex of any Eligible Credit Support, Equivalent Credit Support, Interest Amount or Equivalent Distributions shall be made in accordance with the instructions of the Transferee or Transferor, as applicable, and shall be made:
 - (i) in the case of cash, by transfer into one or more bank accounts specified by the recipient;

- (ii) in the case of certificated securities which cannot or which the parties have agreed will not be delivered by book-entry, by delivery in appropriate physical form to the recipient or its account accompanied by any duly executed instruments of transfer, transfer tax stamps and any other documents necessary to constitute a legally valid transfer of the transferring party's legal and beneficial title to the recipient; and
- (iii) in the case of securities which the parties have agreed will be delivered by book-entry, by the giving of written instructions (including, for the avoidance of doubt, instructions given by telex, facsimile transmission or electronic messaging system) to the relevant depository institution or other entity specified by the recipient, together with a written copy of the instructions to the recipient, sufficient, if complied with, to result in a legally effective transfer of the transferring party's legal and beneficial title to the recipient.

Subject to Paragraph 4 and unless otherwise specified, if a demand for the transfer of Eligible Credit Support or Equivalent Credit Support is received by the Notification Time, then the relevant transfer will be made not later than the close of business on the Settlement Day relating to the date such demand is received; if a demand is received after the Notification Time, then the relevant transfer will be made not later than the close of business on the Settlement Day relating to the day after the date such demand is received.

(b) *Calculations*. All calculations of Value and Exposure for purposes of Paragraphs 2 and 4(a) will be made by the relevant Valuation Agent as of the relevant Valuation Time. The Valuation Agent will notify each party (or the other party, if the Valuation Agent is a party) of its calculations not later than the Notification Time on the Local Business Day following the applicable Valuation Date (or, in the case of Paragraph 4(a), following the date of calculation).

(c) Exchanges.

- (i) Unless otherwise specified in Paragraph 11, the Transferor may on any Local Business Day by notice inform the Transferee that it wishes to transfer to the Transferee Eligible Credit Support specified in that notice (the "New Credit Support") in exchange for certain Eligible Credit Support (the "Original Credit Support") specified in that notice comprised in the Transferor's Credit Support Balance.
- (ii) If the Transferee notifies the Transferor that it has consented to the proposed exchange, (A) the Transferor will be obliged to transfer the New Credit Support to the Transferee on the first Settlement Day following the date on which it receives notice (which may be oral telephonic notice) from the Transferee of its consent and (B) the Transferee will be obliged to transfer to the Transferor Equivalent Credit Support in respect of the Original Credit Support not later than the Settlement Day following the date on which the Transferee receives the New Credit Support, unless otherwise specified in Paragraph 11(d) (the "Exchange Date"); provided that the Transferee will only be obliged to transfer Equivalent Credit Support with a Value as of the date of transfer as close as practicable to, but in any event not more than, the Value of the New Credit Support as of that date.

Paragraph 4. Dispute Resolution

- (a) **Disputed Calculations or Valuations**. If a party (a "Disputing Party") reasonably disputes (I) the Valuation Agent's calculation of a Delivery Amount or a Return Amount or (II) the Value of any transfer of Eligible Credit Support or Equivalent Credit Support, then:
 - (1) the Disputing Party will notify the other party and the Valuation Agent (if the Valuation Agent is not the other party) not later than the close of business on the Local Business Day following, in the case of (I) above, the date that the demand is received under Paragraph 2 or, in the case of (II) above, the date of transfer;
 - (2) in the case of (I) above, the appropriate party will transfer the undisputed amount to the other party not later than the close of business on the Settlement Day following the date that the demand is received under Paragraph 2;
 - (3) the parties will consult with each other in an attempt to resolve the dispute; and
 - (4) if they fail to resolve the dispute by the Resolution Time, then:
 - (i) in the case of a dispute involving a Delivery Amount or Return Amount, unless otherwise specified in Paragraph 11(c), the Valuation Agent will recalculate the Exposure and the Value as of the Recalculation Date by:
 - (A) utilising any calculations of that part of the Exposure attributable to the Transactions that the parties have agreed are not in dispute;
 - (B) calculating that part of the Exposure attributable to the Transactions in dispute by seeking four actual quotations at mid-market from Reference Market-makers for purposes of calculating Market Quotation, and taking the arithmetic average of those obtained; *provided* that if four quotations are not available for a particular Transaction, then fewer than four quotations may be used for that Transaction, and if no quotations are available for a particular Transaction, then the Valuation Agent's original calculations will be used for the Transaction; and
 - (C) utilising the procedures specified in Paragraph 11(e)(ii) for calculating the Value, if disputed, of the outstanding Credit Support Balance;
 - (ii) in the case of a dispute involving the Value of any transfer of Eligible Credit Support or Equivalent Credit Support, the Valuation Agent will recalculate the Value as of the date of transfer pursuant to Paragraph 11(e)(ii).

Following a recalculation pursuant to this Paragraph, the Valuation Agent will notify each party (or the other party, if the Valuation Agent is a party) as soon as possible but in any event not later than the Notification Time on the Local Business Day following the Resolution Time. The appropriate party will, upon demand following such notice given by the Valuation Agent or a resolution pursuant to (3) above and subject to Paragraph 3(a), make the appropriate transfer.

(b) **No Event of Default.** The failure by a party to make a transfer of any amount which is the subject of a dispute to which Paragraph 4(a) applies will not constitute an Event of Default for as long as the procedures set out in this Paragraph 4 are being carried out. For the avoidance of doubt, upon completion of those procedures, Section 5(a)(i) of this Agreement will apply to any failure by a party to make a transfer required under the final sentence of Paragraph 4(a) on the relevant due date.

Paragraph 5. Transfer of Title, No Security Interest, Distributions and Interest Amount

- (a) *Transfer of Title*. Each party agrees that all right, title and interest in and to any Eligible Credit Support, Equivalent Credit Support, Equivalent Distributions or Interest Amount which it transfers to the other party under the terms of this Annex shall vest in the recipient free and clear of any liens, claims, charges or encumbrances or any other interest of the transferring party or of any third person (other than a lien routinely imposed on all securities in a relevant clearance system).
- (b) **No Security Interest.** Nothing in this Annex is intended to create or does create in favour of either party any mortgage, charge, lien, pledge, encumbrance or other security interest in any cash or other property transferred by one party to the other party under the terms of this Annex.

(c) Distributions and Interest Amount.

- (i) *Distributions*. The Transferee will transfer to the Transferor not later than the Settlement Day following each Distributions Date cash, securities or other property of the same type, nominal value, description and amount as the relevant Distributions ("Equivalent Distributions") to the extent that a Delivery Amount would not be created or increased by the transfer, as calculated by the Valuation Agent (and the date of calculation will be deemed a Valuation Date for this purpose).
- (ii) *Interest Amount.* Unless otherwise specified in Paragraph 11(f)(iii), the Transferee will transfer to the Transferor at the times specified in Paragraph 11(f)(ii) the relevant Interest Amount to the extent that a Delivery Amount would not be created or increased by the transfer, as calculated by the Valuation Agent (and the date of calculation will be deemed a Valuation Date for this purpose).

Paragraph 6. Default

If an Early Termination Date is designated or deemed to occur as a result of an Event of Default in relation to a party, an amount equal to the Value of the Credit Support Balance, determined as though the Early Termination Date were a Valuation Date, will be deemed to be an Unpaid Amount due to the Transferor (which may or may not be the Defaulting Party) for purposes of Section 6(e). For the avoidance of doubt, if Market Quotation is the applicable payment measure for purposes of Section 6(e), then the Market Quotation determined under Section 6(e) in relation to the Transaction constituted by this Annex will be deemed to be zero, and, if Loss is the applicable payment measure for purposes of Section 6(e), then the Loss determined under Section 6(e) in relation to the Transaction will be limited to the Unpaid Amount representing the Value of the Credit Support Balance.

Paragraph 7. Representation

Each party represents to the other party (which representation will be deemed to be repeated as of each date on which it transfers Eligible Credit Support, Equivalent Credit Support or Equivalent Distributions) that it is the sole owner of or otherwise has the right to transfer all Eligible Credit Support, Equivalent Credit Support or Equivalent Distributions it transfers to the other party under this Annex, free and clear of any security interest, lien, encumbrance or other restriction (other than a lien routinely imposed on all securities in a relevant clearance system).

Paragraph 8. Expenses

Each party will pay its own costs and expenses (including any stamp, transfer or similar transaction tax or duty payable on any transfer it is required to make under this Annex) in connection with performing its obligations under this Annex, and neither party will be liable for any such costs and expenses incurred by the other party.

Paragraph 9. Miscellaneous

- (a) **Default Interest.** Other than in the case of an amount which is the subject of a dispute under Paragraph 4(a), if a Transferee fails to make, when due, any transfer of Equivalent Credit Support, Equivalent Distributions or the Interest Amount, it will be obliged to pay the Transferor (to the extent permitted under applicable law) an amount equal to interest at the Default Rate multiplied by the Value on the relevant Valuation Date of the items of property that were required to be transferred, from (and including) the date that the Equivalent Credit Support, Equivalent Distributions or Interest Amount were required to be transferred to (but excluding) the date of transfer of the Equivalent Credit Support, Equivalent Distributions or Interest Amount. This interest will be calculated on the basis of daily compounding and the actual number of days elapsed.
- (b) Good Faith and Commercially Reasonable Manner. Performance of all obligations under this Annex, including, but not limited to, all calculations, valuations and determinations made by either party, will be made in good faith and in a commercially reasonable manner.
- (c) *Demands and Notices*. All demands and notices given by a party under this Annex will be given as specified in Section 12 of this Agreement.
- (d) Specifications of Certain Matters. Anything referred to in this Annex as being specified in Paragraph 11 also may be specified in one or more Confirmations or other documents and this Annex will be construed accordingly.

Paragraph 10. Definitions

As used in this Annex:

"Base Currency" means the currency specified as such in Paragraph 11(a)(i).

"Base Currency Equivalent" means, with respect to an amount on a Valuation Date, in the case of an amount denominated in the Base Currency, such Base Currency amount and, in the case of an amount denominated in a currency other than the Base Currency (the "Other Currency"), the amount of Base Currency required to purchase such amount of the Other Currency at the spot exchange rate determined by the Valuation Agent for value on such Valuation Date.

"Credit Support Amount" means, with respect to a Transferor on a Valuation Date, (i) the Transferee's Exposure plus (ii) all Independent Amounts applicable to the Transferor, if any, minus (iii) all Independent Amounts applicable to the Transferee, if any, minus (iv) the Transferor's Threshold; provided, however, that the Credit Support Amount will be deemed to be zero whenever the calculation of Credit Support Amount yields a number less than zero.

"Credit Support Balance" means, with respect to a Transferor on a Valuation Date, the aggregate of all Eligible Credit Support that has been transferred to or received by the Transferee under this Annex, together with any Distributions and all proceeds of any such Eligible Credit Support or Distributions, as reduced pursuant to Paragraph 2(b), 3(c)(ii) or 6. Any Equivalent Distributions or Interest Amount (or portion of either) not transferred pursuant to Paragraph 5(c)(i) or (ii) will form part of the Credit Support Balance.

"Delivery Amount" has the meaning specified in Paragraph 2(a).

"Disputing Party" has the meaning specified in Paragraph 4.

"Distributions" means, with respect to any Eligible Credit Support comprised in the Credit Support Balance consisting of securities, all principal, interest and other payments and distributions of cash or other property to which a holder of securities of the same type, nominal value, description and amount as such Eligible Credit Support would be entitled from time to time.

"Distributions Date" means, with respect to any Eligible Credit Support comprised in the Credit Support Balance other than cash, each date on which a holder of such Eligible Credit Support is entitled to receive Distributions or, if that date is not a Local Business Day, the next following Local Business Day.

"Eligible Credit Support" means, with respect to a party, the items, if any, specified as such for that party in Paragraph 11(b)(ii) including, in relation to any securities, if applicable, the proceeds of any redemption in whole or in part of such securities by the relevant issuer.

"Eligible Currency" means each currency specified as such in Paragraph 11(a)(ii), if such currency is freely available.

"Equivalent Credit Support" means, in relation to any Eligible Credit Support comprised in the Credit Support Balance, Eligible Credit Support of the same type, nominal value, description and amount as that Eligible Credit Support.

"Equivalent Distributions" has the meaning specified in Paragraph 5(c)(i).

"Exchange Date" has the meaning specified in Paragraph 11(d).

"Exposure" means, with respect to a party on a Valuation Date and subject to Paragraph 4 in the case of a dispute, the amount, if any, that would be payable to that party by the other party (expressed as a positive number) or by that party to the other party (expressed as a negative number) pursuant to Section 6(e)(ii)(1) of this Agreement if all Transactions (other than the Transaction constituted by this Annex) were being terminated as of the relevant Valuation Time, on the basis that (i) that party is not the Affected Party and (ii) the Base Currency is the Termination Currency; provided that Market Quotations will be determined by the Valuation Agent on behalf of that party using its estimates at mid-market of the amounts that would be paid for Replacement Transactions (as that term is defined in the definition of "Market Quotation").

"Independent Amount" means, with respect to a party, the Base Currency Equivalent of the amount specified as such for that party in Paragraph 11 (b)(iii)(A); if no amount is specified, zero.

"Interest Amount" means, with respect to an Interest Period, the aggregate sum of the Base Currency Equivalents of the amounts of interest determined for each relevant currency and calculated for each day in that Interest Period on the principal amount of the portion of the Credit Support Balance comprised of cash in such currency, determined by the Valuation Agent for each such day as follows:

- (x) the amount of cash in such currency on that day; multiplied by
- (y) the relevant Interest Rate in effect for that day; divided by
- (z) 360 (or, in the case of pounds sterling, 365).

"Interest Period" means the period from (and including) the last Local Business Day on which an Interest Amount was transferred (or, if no Interest Amount has yet been transferred, the Local Business Day on which Eligible Credit Support or Equivalent Credit Support in the form of cash was transferred to or received by the Transferee) to (but excluding) the Local Business Day on which the current Interest Amount is transferred.

"Interest Rate" means, with respect to an Eligible Currency, the rate specified in Paragraph 11(f)(i) for that currency.

"Local Business Day", unless otherwise specified in Paragraph 11(h), means:

- (i) in relation to a transfer of cash or other property (other than securities) under this Annex, a day on which commercial banks are open for business (including dealings in foreign exchange and foreign currency deposits) in the place where the relevant account is located and, if different, in the principal financial centre, if any, of the currency of such payment;
- (ii) in relation to a transfer of securities under this Annex, a day on which the clearance system agreed between the parties for delivery of the securities is open for the acceptance and execution of settlement instructions or, if delivery of the securities is contemplated by other means, a day on which commercial banks are open for business (including dealings in foreign exchange and foreign currency deposits) in the place(s) agreed between the parties for this purpose;

- (iii) in relation to a valuation under this Annex, a day on which commercial banks are open for business (including dealings in foreign exchange and foreign currency deposits) in the place of location of the Valuation Agent and in the place(s) agreed between the parties for this purpose; and
- (iv) in relation to any notice or other communication under this Annex, a day on which commercial banks are open for business (including dealings in foreign exchange and foreign currency deposits) in the place specified in the address for notice most recently provided by the recipient.

"Minimum Transfer Amount" means, with respect to a party, the amount specified as such for that party in Paragraph 11(b)(iii)(C); if no amount is specified, zero.

"New Credit Support" has the meaning specified in Paragraph 3(c)(i).

"Notification Time" has the meaning specified in Paragraph 11(c)(iv).

"Recalculation Date" means the Valuation Date that gives rise to the dispute under Paragraph 4; provided, however, that if a subsequent Valuation Date occurs under Paragraph 2 prior to the resolution of the dispute, then the "Recalculation Date" means the most recent Valuation Date under Paragraph 2.

"Resolution Time" has the meaning specified in Paragraph 11(c)(i).

"Return Amount" has the meaning specified in Paragraph 2(b).

"Settlement Day" means, in relation to a date, (i) with respect to a transfer of cash or other property (other than securities), the next Local Business Day and (ii) with respect to a transfer of securities, the first Local Business Day after such date on which settlement of a trade in the relevant securities, if effected on such date, would have been settled in accordance with customary practice when settling through the clearance system agreed between the parties for delivery of such securities or, otherwise, on the market in which such securities are principally traded (or, in either case, if there is no such customary practice, on the first Local Business Day after such date on which it is reasonably practicable to deliver such securities).

"Threshold" means, with respect to a party, the Base Currency Equivalent of the amount specified as such for that party in Paragraph 11(b)(iii)(B); if no amount is specified, zero.

"Transferee" means, in relation to each Valuation Date, the party in respect of which Exposure is a positive number and, in relation to a Credit Support Balance, the party which, subject to this Annex, owes such Credit Support Balance or, as the case may be, the Value of such Credit Support Balance to the other party.

"Transferor" means, in relation to a Transferee, the other party.

"Valuation Agent" has the meaning specified in Paragraph 11(c)(i).

"Valuation Date" means each date specified in or otherwise determined pursuant to Paragraph 11(c)(ii).

"Valuation Percentage" means, for any item of Eligible Credit Support, the percentage specified in Paragraph 11(b)(ii).

"Valuation Time" has the meaning specified in Paragraph 11(c)(iii).

"Value" means, for any Valuation Date or other date for which Value is calculated, and subject to Paragraph 4 in the case of a dispute, with respect to:

- (i) Eligible Credit Support comprised in a Credit Support Balance that is:
 - (A) an amount of cash, the Base Currency Equivalent of such amount multiplied by the applicable Valuation Percentage, if any; and
 - (B) a security, the Base Currency Equivalent of the bid price obtained by the Valuation Agent multiplied by the applicable Valuation Percentage, if any; and
- (ii) items that are comprised in a Credit Support Balance and are not Eligible Credit Support, zero.

Paragraph 11. Elections and Variables

(a)	Base	Currency and Eligible currency.			
	(i)	"Base Currency" means United States Dollars unle	ess otherwise s	specified here:	
	(ii)	"Eligible Currency" means the Base Currency and	each other cu	rrency specifie	d here:
(b)	Credi	t Support Obligations.			
	(i)	Delivery Amount, Return Amount and Credit Sup	pport Amount	•	
		(A) "Delivery Amount" has the meaning spec otherwise specified here:			
		(B) "Return Amount" has the meaning speci specified here:			s otherwise
		(C) "Credit Support Amount" has the meaning otherwise specified here:			unless
	(ii) for th	Eligible Credit Support . The following items will e party specified:	l qualify as '' I	Eligible Credit	Support''
			Party A	Party B	Valuation Percentage
	(A)	cash in an Eligible Currency	[]	[]	[]%
	(B)	negotiable debt obligations issued by the Government of [] having an original maturity at issuance of not more than one year		[]	[]%
	(C)	negotiable debt obligations issue by the Government of [] having an original maturity at issuance of more than one year but not more than 10 years		[]	[]%

	(D)	negotiable debt obligation issued by the [] [] []% Government of [] having an original maturity at issuance of more than 10 years
	(E)	other: [] []%
	(iii)	Thresholds.
	(A)	"Independent Amount" means with respect to Party A: "Independent Amount" means with respect to Party B:
	(B)	"Threshold" means with respect to Party A: "Threshold" means with respect to Party B:
	(C)	"Minimum Transfer Amount" means with respect to Party A: "Minimum Transfer Amount" means with respect to Party B:
	(D) neares	Rounding. The Delivery Amount and the Return Amount will be rounded [down to the st integral multiple of, respectively ³].
(c)	Valua	tion and Timing.
		"Valuation Agent" means, for purposes of Paragraphs 2 and 4, the party making the nd under Paragraph 2, and, for purposes of Paragraph 5(c), the Transferee, as applicable, otherwise specified here:
	(ii)	"Valuation Date" means:
	(iii)	"Valuation Time" means:
		[] the close of business in the place of location of the Valuation Agent on the Valuation Date or date of calculation, as applicable;
		[] the close of business on the Local Business Day immediately preceding the Valuation Date or date of calculation, as applicable;
	-	ded that the calculations of Value and Exposure will, as far as practicable, be made as of ximately the same time on the same date.
	(iv) otherw	"Notification Time" means 1:00 p.m., London time, on a Local Business Day, unless vise specified here:
³ Delet	te as appl	icable.

Dispute Reso	lution.	
the date on	-	don time, on the Local Business Day following s rise to a dispute under Paragraph 4, unless
outstanding	Credit Support Balance or of any tra	H(a)(4)(i)(C) and 4(a)(4)(ii), the Value of the nsfer of Eligible Credit Support or Equivalent ated as follows:
		4 will apply, unless an alternative dispute
Distributions	and Interest Amount.	
(i) Interwill		on to each Eligible Currency specified below
	Eligible Currency	Interest Rate
last Local B Amount cor Paragraph 20	usiness Day of each calendar month asisting wholly or partly of cash (b), unless otherwise specified here:	r of the Interest Amount will be made on the and on any Local Business Day that a Return is transferred to the Transferor pursuant to
last Local B Amount cor Paragraph 20 (iii) Alter	usiness Day of each calendar month asisting wholly or partly of cash (b), unless otherwise specified here:	r of the Interest Amount will be made on the and on any Local Business Day that a Return is transferred to the Transferor pursuant to risions of Paragraph 5(c)(ii) will apply, unless
last Local B Amount cor Paragraph 20 (iii) Alter	usiness Day of each calendar month asisting wholly or partly of cash (b), unless otherwise specified here:	r of the Interest Amount will be made on the and on any Local Business Day that a Return is transferred to the Transferor pursuant to risions of Paragraph 5(c)(ii) will apply, unless
last Local B Amount cor Paragraph 2(iii) Alter otherwise spe	usiness Day of each calendar month asisting wholly or partly of cash (b), unless otherwise specified here:	r of the Interest Amount will be made on the and on any Local Business Day that a Return is transferred to the Transferor pursuant to risions of Paragraph 5(c)(ii) will apply, unless

SCHEDULE 1 - CREDIT SUPPORT ANNEX PARAGRAPH 11

Paragraph 11. Elections and Variables

- (a) Base Currency and Eligible Currency.
 - (i) "Base Currency" means Australian Dollars.
 - (ii) "Eligible Currency" means the Base Currency.
- (b) Credit Support Obligations.
 - (i) Delivery Amount, Return Amount and Credit Support Amount.
 - (A) "Delivery Amount"

Party A is not required to deliver any Eligible Credit Support (whether by way of Delivery Amounts or Return Amounts or otherwise) to Party B in respect of the Transaction covered by this Confirmation.

Despite any other provision of this Confirmation, Party B must on the Initial Exchange Date deliver to Party A an initial Delivery Amount comprised of Shares in an amount equal to the Aggregate Number of Shares. Party A agrees that it will not, for the purposes of the Subscription Agreement, direct the Issuer to register Party B or UBS Nominees Pty Limited as nominee of Party B as the holder of Shares in an amount other than the Aggregate Number of Shares. No other Delivery Amounts are required to be delivered by Party B to Party A except:

- (aa) in circumstances where Eligible Credit Support in the form of Shares has been transferred by Party A to the Nominee in accordance with "Additional Provisions paragraph 4 Voting Arrangements" of this Confirmation, in which case the required Delivery Amount is equivalent to the number of Shares so transferred to the Nominee. The Delivery Amount must be delivered to Party A in accordance with "Additional Provisions paragraph 4 Voting Arrangements" of this Confirmation and the Nominee Deed; and
- (bb) in circumstances where Party A has agreed to the delivery or substitution of collateral other than the Shares, in which case any required Delivery Amounts are determined in accordance with the agreement between Party A and Party B in respect of that alternative collateral.

(B) "Return Amount"

Despite any other provision of this Confirmation, Party A must on the Cash Settlement Payment Date (if Cash Settlement applies) or Settlement Date (if Physical Settlement applies) for a Tranche, deliver a Return Amount comprised of Shares in an amount equal to the Tranche Quantity for that Tranche to Party B, provided that no Event of Default (as defined in the ISDA Form) has occurred in respect of which Party B is the Defaulting Party or Termination Event has occurred in respect of which Party B is the Affected Party.

To the extent that, on a Settlement Date, Party A is required to deliver a Return Amount to Party B (such quantity of Shares being "Quantity X") and Party B is on that date required to deliver Shares to Party A in connection with the physical settlement of the Related Options (such quantity of Shares being "Quantity Y"), Party A's obligation to deliver Shares under this Confirmation in respect of the

relevant Tranche shall be satisfied by the delivery of a number of Shares to Party B on the Settlement Date for that Tranche, equal to the excess, if any, of Quantity X over Quantity Y.

- (C) "Credit Support Amount" in Paragraph 10 does not apply and, instead, for the purposes of this Confirmation, it means at any time an amount equal to the Aggregate Number of Shares comprised in the Delivery Amount on the Initial Exchange Date less any Shares delivered to Party B under paragraph (B) ("Return Amount") above.
- (ii) Eligible Credit Support. The following items will qualify as "Eligible Credit Support":

	Party B	Valuation Percentage
Fully paid ordinary shares of the Issuer, or any other collateral agreed to in advance in writing by Party A	✓	100%
Australian Dollars cash	✓	100%

"Issuer" means Oil Search Limited (ARBN 055 079 868). ASX code "OSH".

- (iii) Thresholds.
 - (A) "Independent Amount" means with respect to Party A: zero "Independent Amount" means with respect to Party B: zero
 - (B) "Threshold" means zero for both Party A and party B.
 - (C) "Minimum Transfer Amount" means AUD 1,000,000 for both Party A and Party B.
 - (D) Rounding. AUD 10,000
- (c) Valuation and Timing.
 - (i) "Valuation Agent" means Party A.
 - (ii) "Valuation Date" means any Local Business Day.
 - (iii) "Valuation Time" means the close of business in the Relevant Market on the Local Business Day first preceding the Valuation Date or date of calculation, as applicable.

For the purposes of this provision, "Relevant Market" means:

- (a) with respect to the calculation of Value, the principal market in which the Eligible Credit Support is traded; and
- (b) with respect to the calculation of Exposure, the location most closely associated with the relevant Transaction;

each as determined by the Valuation Agent, or as otherwise agreed between the parties.

(iv) "Notification Time" means 3:00 p.m., Sydney time, on the Local Business Day.

- (d) Exchange Date. "Exchange Date" has the meaning specified in Paragraph 3(c)(ii).
- (e) Dispute Resolution.
 - (i) "Resolution Time" means 3:00 p.m., Sydney time, on the Local Business Day following the date on which the notice is given that gives rise to a dispute under Paragraph 4.
 - (ii) Value. For the purpose of Paragraphs 4(a)(4)(i)(C) and 4(a)(4)(ii), , disputes over value will be resolved by the Valuation Agent seeking three bid quotes as of the relevant Valuation Date or date of Transfer from parties that regularly act as dealers in the securities or other property in question. The Value will be the Base Currency Equivalent of the arithmetic mean of the bid prices obtained by the Valuation Agent multiplied by the nominal amount of such security, plus any income which, as of such date, has accrued but not yet been paid in respect of the security to the extent not included in such price as of such date, multiplied by the applicable Valuation Percentage.
 - (iii) Alternative. The provisions of Paragraph 4 will apply.
- (f) Distributions and Interest Amount.

Despite anything to the contrary in the 1995 ISDA Credit Support Annex forming part of the ISDA Form, Party A is only required to pay "Equivalent Distributions" to Party B as contemplated under paragraph 5(c) of the 1995 ISDA Credit Support Annex as follows:

- a) If the ex-date of an Extraordinary Dividend declared in respect of the Shares occurs during the Dividend Period then Party B agrees that:
 - (i) Party A shall not be required to pay an Equivalent Distribution to Party B in respect of a quantity of Shares equal to the Delta Quantity;
 - (ii) Party A shall pay to Party B an Equivalent Distribution equal to the Actual Dividend Amount per Share in respect of the Non-Delta Quantity; and
 - (iii) payment under paragraph (ii) shall be made by no later than the second Business Day after the payment date of such Extraordinary Dividend or cash return by the Issuer.
- b) If the ex-date of an ordinary dividend declared in respect of the Shares occurs during the Dividend Period then Party B agrees that:
 - (i) Party A shall pay to Party B an Equivalent Distribution equal to A\$0.022 per Share in respect of the Delta Quantity;
 - (ii) Party A shall pay to Party B an Equivalent Distribution equal to the Actual Dividend Amount per Share in respect of the Non-Delta Quantity; and
 - (iii) payment under paragraphs (i) and (ii) shall be made by no later than the second Business Day after the payment date of such ordinary dividend by the Issuer.

No Interest Amounts are payable by Party A to Party B.

For the purposes of the above:

"Delta Quantity" is a number determined by UBS that is between zero and the Aggregate Number of Shares.

"Non-Delta Quantity" is calculated as the Aggregate Number of Shares minus the Delta Quantity.

"Actual Dividend Amount" is the Australian dollar dividend amount per Share declared by the Issuer, or if the Issuer does not declare a dividend in Australian dollars then it is the

dividend amount per share declared by the Issuer multiplied by an exchange rate determined by the Calculation Agent on the ex-dividend date to convert that amount into Australian dollars, less any withholding that would be applicable to Party A when it makes a payment to Party B under this Confirmation, and less any withholding that would be applicable to a shareholder in the same position and with the same characteristics as Party A.

"Dividend Period" means the period from but excluding the Trade Date, to and including the Expiration Date for the relevant Component Collar.

(g) Demands and Notices / Address for Transfers.

(i) Any demand, specification or notice under this Annex (each, a "Notice") must be delivered in writing which, for the avoidance of doubt, shall include electronic messaging or email. All Notices shall be delivered to the following email addresses and physical addresses:

With respect of Party A:

Level 16 Chifley Tower 2 Chifley Square Sydney NSW 2000 Australia

Email address: elaine.chow@ubs.com / kevin.bayett@ubs.com / sh-syd-rmp-op@ubs.com Telephone: +61 2 9324 2216 / +61 2 9324 3140 / +61 2 9324 3193

Fax number: +61-2-9324-2880

Attention: Elaine Chow / Kevin Bayett / Michael Rollin

With respect to Party B:

The Independent State of Papua New Guinea Department of Treasury P O Box 542 Waigani 131 N.C.D. Papua New Guinea

Email address: dairi_vele@treasury.gov.pg

Telephone:

+675 3235600

Fax Number:

+675 3128804

Attention:

Mr Dairi Vele, Secretary for Department of Treasury

(ii) Addresses and account details for Transfers.

With respect of Party A:

UBS AG, Australia Branch

BSB: 946-612 Account#: 242888

Attn: Collateral Mgr Sydney Exigo Code: UBSB20 For delivery of Shares:

PID: 1505

HIN: X0000103462

With respect to Party B:

To be advised by Party B to Party A

(h) Other Provisions.

This Annex, including the act of transfer (disposition) itself, will be governed and construed in accordance with New South Wales law.

Holder of relevant interest	UBS AG, Australia Branch
Type of agreement	Prime Brokerage Agreement
Parties to agreement	(i) UBS AG, Australia Branch ("UBS AG") (ii) Client (Please refer to Appendix A-1.)
Transfer date	Please refer to Annexure E-1.
Holder of voting rights	UBS AG
Are there any restrictions on voting rights?	Please refer to the details below.
If yes, detail	Since all right and title in the securities passes to the transferee (i.e. UBS AG), the transferee has the right to vote. However, in certain circumstances, the transferee may seek to arrange for instructions to be exercised in accordance with the instructions of the transferor (i.e. Client).
Scheduled return date (if any)	None, subject to the terms of the relevant loan.
Does the borrower have the right to return early?	Yes.
If yes, detail	The borrower (i.e. UBS AG) has the right to return at its discretion.
Does the lender have the right to recall early?	Yes.
If yes, detail	The lender (i.e. Client) can recall at its request subject to compliance with margin requirements and the terms of the relevant loan.
Will the securities be returned on settlement?	Yes.
If yes, detail any exceptions	None.

Prime Brokerage Agreement - UBS AG, Australia Branch

Parties to agreement	Transfer date
Australian Leaders Fund Limited	(i) 14 March 2014
	(ii) 17 March 2014
	(iii) 18 March 2014
	(iv) 10 April 2014
	(v) 22 April 2014
Perpetual Investment Management Limited as responsible entity of Perpetual's	(i) 14 March 2014
Wholesale Geared Australian Fund	(ii) 17 March 2014
	(iii) 18 March 2014
	(iv) 20 March 2014
	(v) 27 March 2014
	(vi) 10 April 2014
	(vii) 14 April 2014
	(viii) 30 April 2014
	(ix) 11 July 2014
Regal Funds Management Pty Ltd as trustee and manager of Atlantic Absolute Return	(i) 14 March 2014
Fund	(ii) 17 March 2014
	(iii) 25 March 2014
	(iv) 9 April 2014
	(v) 22 April 2014
	(vi) 28 April 2014
	(vii) 29 April 2014
	(viii) 7 May 2014
	(ix) 16 May 2014
	(x) 20 May 2014
	(xi) 21 May 2014
	(xii) 25 June 2014
Bennelong Funds Management Pty Ltd as trustee of Bennelong Long Short Equity Fund	(i) 14 March 2014
	(ii) 17 March 2014
	(iii) 19 March 2014
	(iv) 25 March 2014
	(v) 7 May 2014
	(vi) 15 May 2014

Annexure E-1

Regal Funds Management Pty Limited as trustee for Regal Australian Long Short	(i) 14 March 2014
Geared Equity Fund	(ii) 17 March 2014
	(iii) 19 March 2014
	(iv) 9 April 2014
	(v) 30 April 2014
	(vi) 1 May 2014
	(vii) 2 May 2014
	(viii) 16 May 2014
Watermark Funds Management Pty Limited as responsible entity for Watermark Market	(i) 14 March 2014
Neutral Fund	(ii) 17 March 2014
	(iii) 19 June 2014
	(iv) 23 June 2014
Atrium Investment Management Pty Ltd in its capacity as trustee of the scheme Atrium	(i) 14 March 2014
Benn L-S Fund	(ii) 17 March 2014
	(iii) 19 March 2014
	(iv) 25 March 2014
Watermark Market Neutral Fund Limited	(i) 14 March 2014
	(ii) 21 March 2014
	(iii) 24 March 2014
	(iv) 9 April 2014
Bennelong Funds Management Ltd as Trustee for Bennelong Alpha 200 Fund	(i) 14 March 2014
	(ii) 16 April 2014
	(iii) 22 April 2014
Regal Funds Management Pty Limited as trustee and manager of Tasman Market	(i) 17 March 2014
Neutral Fund	(ii) 10 July 2014
· · · · · · · · · · · · · · · · · · ·	
Valuestream Investment Management Limited as Trustee for the Imperia Asia Fund	(i) 17 March 2014

			in relation to change	securities	
3-Mar-14	UBS AG Australia Branch	Self	50,475	(5,844)	Ordinary
3-Mar-14	UBS AG Australia Branch	Buy	960,604	111,488	Ordinary
-Mar-14	Г	Sell	2,199,385	(254,473)	Ordinary
3-Mar-14	UBS AG	Buv	2 225 481	254.473	Ordinary
3-Mar-14	Т	Stock borrowed	N/A	250,000	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Stock borrowed	AVA	82,000	Ordinary
-Mar-14	UBS Securities Australia Ltd	Stock borrowed	A/N	20,000	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Stock borrowed	NA	6,324	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Stock borrowed	ΑN	5.844	Ordinary
3-Mar-14	UBS Securities Australia I td	Stock borrowed	A/A	2000	Ordinary
3-Mar-14	UBS Securities Australia I to	Stock horroward	VIV	255,000	Ordinan
3-Mar-14	I IBS Securities Australia 14	Stock returned		000,000,00	y in the contract of the contr
Af ac 1 A	TIPO Countries Australia 144	Stock returned	¥2.	(000,000,00)	Cidinary
		Slock returned	YA.	(20,000,000)	Ordinary
i i		Stock returned	N.	(000,000,c)	Ordinary
D-IMAI-14	UBS AG LONGON Branch	Stock returned	Y.	(1,300,000)	Ordinary
3-Mar-14		Stock returned	ΥN	(1,000,000)	Ordinary
3-Mar-14		Stock returned	ΝA	(1,300,000)	Ordinary
3-Mar-14		Stock returned	N/A	(2,000,000)	Ordinary
Mar-14	UBS AG London Branch	Stock returned	ΑN	(8,000,000)	Ordinary
3-Mar-14	UBS AG London Branch	Stock returned	ΑN	(3,000,000)	Ordinary
3-Mar-14	UBS AG London Branch	Stock returned	₹ 2	(140,000)	Ordinary
3-Mar-14	UBS AG London Branch	Stock returned	AW	(160.000)	Ordinary
13-Mar-14		Stock returned	AVA	(5,000,000)	Ordinary
Mar-14		Stock returned	ANN	(5,000,000)	Ordinan
3-Mar-14		Stock returned	V.	(100,000,000)	Ordinary
2 Mar 1 A		Stock letering	29	(000,001)	Clarialy
IN OIL		Stock returned	Y N	(100,000)	Ordinary
3-Mar-14	AG London Branch	Stock returned	∀N V	(3,000,000)	Ordinary
3-Mar-14	AG London Branch	Stock returned	NA	(940,000)	Ordinary
Mar-14	Ltd	Sell	2,934	(339)	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Sell	43.644	(5:005)	Ordinary
3-Mar-14		Riv	43 859	2 050	Ordinan
3-Mar-14		7.0	260,000	2000	Control of
2 kd 2 1 A		670	302,340	040,240	Oldinary
	COO Decarities Augustia Liu		790,167	(6//07)	Ordinary
NO.			3///6	(7,787)	Ordinary
3-ivlar-14		Buy	233,916	27,012	Ordinary
3-Mar-14	Securities Australia Ltd	Sell	2,199,385	(254,473)	Ordinary
3-Mar-14		Stock returned	N/A	(3,000,000)	Ordinary
Mar-14		Stock returned	VV	(7 200 000)	Ordinary
3-Mar-14		Stock returned	4714	(4,000,000)	Consister C
		Stock returned	¥N.	(4,000,000)	Ordinary
3-Mar-14		- A	2,255	(292)	Ordinary
3-Mar-14		Buy	178,005	20,477	Ordinary
3-Mar-14		Buv	1.098.957	127 047	Ordinary
Mar-14			LTC 000 '	10,121	, in the second
ti-laivi-Ci	COD GLOBAL ASSET IN ALL ASSET I	Áno	708,080,1	127,047	Orginary
Mar-14		Buy	1,098,957	127,047	Ordinary
3-Mar-14		Buy	1,098,957	127,047	Ordinary
3-Mar-14	UBS Global Asset Management	Buv	1.098 957	127 047	Ordinary
3-Mar-14			1 000 5	120,121	in in it
, all	Diobal Asset Ivial agentient	in a	1,096,950,1	127,047	Ordinary
3-Mar-14		Buy	2,256	292	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	33,080	3,842	Ordinary
3-Mar-14		Buy	2 833	379	Ordinary
3-Mar-14	Securities Australia 14d		10 000	200	
	Securities Australia Etc	5	occ,c1	1,00,1	Ordinary
3-Mar-14		Buy	53,718	6,239	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	27,552	3,200	Ordinary
3-Mar-14		Buv	11.945	1 389	Ordinary
3-Mar-14		Brox	10.031	1771	Ordinan
2 Mar 14	Library Australia Australia		100,01	1,2,1	oldinal y
2 Mar 14		, and	1641	100	Oldinally .
VIGIT-14		Buy	20,416	2,3/4	Ordinary
3-Mar-14		Buy	17,028	1,980	Ordinary
3-Mar-14		Buy	3,182	370	Ordinary
3-Mar-14	1085 Securities Australia Ltd	Silv	1118	130	Ordinary
3-Mar-14		, and	900	203	Ordinan
	The County of th	,	D to co	/00	Oldividiy
5-Mar-14		Buy	31,708	3,687	Ordinary
3-Mar-14		Buy	40.325	4 689	Ordinary
		, d	20101	- 3337	
.Mar-14					(in in)

13-Mar-14	I IBS Countition Australia Ltd				
13-Mar-14	UBS Securities Australia Itd	Buy	10 334	1 203	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	43	5	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	6	-	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	292	34	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	5,452	634	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	19,092	2,220	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	22,386	2,603	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	98,931	11,517	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	23,005	2,675	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	4,747	552	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	3,870	450	Ordinary
13-Mar-14	LIRS Securities Australia Ind	na a	49,751	2,785	Ordinary
13-Mar-14	LIBS Securities Australia Ltd	Buy	120	167'/	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buv	5 736	799	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buv	12,049	1.401	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	1,221	142	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	17,381	2,021	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	29,309	3,408	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	2,141	249	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	40,644	4,726	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	1,428	166	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	327	38	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	3,251	378	Ordinary
13-Mar-14	LIBS Securities Australia Ltd	pu)	4,137	481	Ordinary
13-Mar-14	UBS Securities Australia 114	B ₁	0/2/1	1,4/4	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	202	68	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	78,432	9.120	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	5,814	9/9	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	78,905	9,175	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	13,915	1,618	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
13 Mar 14	Upo Securities Australia Ltd	Buy	84,753	3,855	Ordinary
13-Mar-14	UBS Securities Australia I td	B ₁ ×	19,932	2,320	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	903	105	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	112	13	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	17,828	2,073	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	52)65	6:63	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	3,681	428	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	17	2	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	1,729	107	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	30.272	3,520	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	52	9	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	38/	2 45	Ordinary
	UBS Securities Australia Itd	Buy	0,001	2,191	Ordinary
1	UBS Securities Australia Ltd	Buy	0	- -	Ordinary
	UBS Securities Australia Ltd	Buy	129	15	Ordinary
1 1	UBS Securities Australia Ltd	Buy	506	24	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	9,752	1,134	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	1,127	131	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	13,769	1,601	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Bity	120	, ń	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	26	j m	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	2,778	323	Ordinary

13-Mar-14	LIRS Securities Australia I to	N. W.	257	24	Ordina
14-Mar-14	UBS AG Australia Branch	Bliv	256 923	30,000	Ordinary Ordinary
14-Mar-14	UBS AG Australia Branch	Buy	599.488	70,000	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Stock borrowed	A/N	2,000	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Stock returned	A/A	(10,000,000)	Ordinary
14-Mar-14	UBS AG London Branch	Stock borrowed	AVA	3 200 000	Ordinan
14-Mar-14	UBS AG London Branch	Stock returned	A/N	(200,000)	Ordinary
14-Mar-14	UBS AG London Branch	Stock returned	AVA	(1 599 086)	Ordinary
14-Mar-14	UBS AG London Branch	Stock returned	AVA	(1 600 914)	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buv	38.843	4 528	Ordinary
14-Mar-14	UBS Securities Australia Itd	العال	38.845	(0.570)	Ordings
14-Mar-14	I IRS Securities Australia 14d	200	12 001	10/0/0	Ordiciary
14-Mar-14	LIRS Societies Australia 14d	600	010,400	7,300	Ordinary
1 A Mar 1 A	LIBS Complete Australia Liu	and some	010,402	95,780	Ordinary
14-ividi-14	UBS Securities Australia Ltd	Sell	1,226,188	(143,273)	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Sell	119,910	(14,040)	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	57,457	669'9	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	152,888	17,841	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Sell	45,588	(5,291)	Ordinary
14-Mar-14	UBS Global Asset Management	Buy	168,383	19,618	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	17,789	2,081	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	110,011	2,224	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	4,257	498	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	23,858	2,791	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	13,669	1,599	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	3,445	403	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	8,839	1.034	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	9,018	1,055	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	12.267	1,435	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	40.707	4.762	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	11,822	1,383	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	16.806	1.966	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	13,900	1,626	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	22,439	2,625	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	13.865	1,622	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buv	12.421	1 453	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buv	13,002	1.521	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	11.617	1 359	Ordinary
14-Mar-14	U8S Wealth Management Australia 11d	Bliv	3000	1161	Ordingo
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	7 275	851	ording)
14-Mar-14	UBS Wealth Management Australia I to	Bury	13 113	1531	Cidinally
14-Mar-14	I IBS Wealth Management Australia Ind	Buy	27 183	1,334	Ordinary
14-Mar-14	Libe Wealth Macacomont Australia 14d	600	13 575	4.090	Ordinary
14 14 14	UBS Wealth Management Australia Ltd	Buy	5/5/51	886,1	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	BUJ	13,823	1,617	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	26,209	3,066	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	12,686	1,484	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	26,765	3,131	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Виу	13,284	1,554	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	13,233	1,548	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Виу	25,038	2,929	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	11,523	1,348	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Виу	34,219	4,003	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	26,132	3,057	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	14,284	1,671	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	8,728	1,021	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	42,742	2,000	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	52,619	2,646	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	22,379	2,618	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	13,523	1,582	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	31,637	3,701	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	20,661	2,417	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	26,970	3,155	Ordinary
14-Mar-14	UBS Wealth Management Australia Ltd	Buy	11,412	1,335	Ordinary
1	UBS Wealth Management Australia Ltd	Buy	3,865	1,154	Ordinary
	UBS Wealth Management Australia Ltd	Buy	9,044	1.058	Ordinary
Γ					(15:15:)
14-Mar-14	IUBS Wealth Management Australia Ltd	Buv	57,205	6.692	Ordinary

2 4425	1 IDC Constitute Attention 1 and		0,000	,	
3-Mar-14	UBS Securities Australia 11d	R _{IIV}	010,50	4,530	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Riv	112	2,300	Ordinary
3-Mar-14	UBS Sequirities Australia I td	Birv	1 960	000	Ordinary.
Aar-14	I IRS Securities Australia Ltd	Sin.	110 61	1 400	Ordinary
3 Mar 14	DO Coourties Australia 144	600	110,23	064,	Courage Courage
3-Mar-14	UBS Sequities Australia I to	Briv	5,635	523	Ordinary
3-Mar-14	1.185 Securities Australia Ltd	Ruy	0.00	17	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	1 152	134	Ordinary
Mar-14	UBS Securities Australia Ltd	Buv	3.156	367	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buv	13,253	1.541	Ordinary
3-Mar-14	UBS Securities Australia Ltd	8uv	24 691	2871	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	5,315	618	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	24,811	2.885	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	318	37	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	4,868	266	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	52,555	6.111	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	138	16	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	13,459	1,565	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	3,182	370	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	413	48	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	600'2	815	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	2,855	332	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	1,926	224	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	38,382	4,463	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	11,524	1,340	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	7,585	882	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	2,829	329	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	826	96	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	189	22	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	25,785	3,000	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Виу	866'6	1,162	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	5,444	633	Ordinary
Mar-14	UBS Securities Australia Ltd	8uy	3,182	370	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	22,291	2,592	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	39,870	4,636	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	10,148	1,180	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	42,708	4,966	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	6	-	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	1,944	226	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	7,491	871	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	10,905	1,268	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	4,558	530	Ordinary
Var-14	UBS Securities Australia Ltd	Buy	16,873	1,962	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	84,177	9,788	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	26,927	3,131	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	48,332	5,620	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	65,145	7,575	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	29,842	3,470	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	3,836	446	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	31,218	3,630	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	16,185	1,882	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	619	72	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	4,567	531	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	3,560	414	Ordinary
Jar-14	UBS Securities Australia Ltd	Buy	229	65	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	378	44	Ordinary
Aar-14	UBS Securities Australia Ltd	Buy	5,031	585	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	6,082	1,056	Ordinary
13-Mar-14	UBS Securities Australia Ltd	Buy	098	100	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	94,617	11,002	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	2,537	295	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	4,016	457	
				101	Crainary
13-Mar-14	UBS Securities Australia Ltd	Buv	198	73 40	Ordinary

4-Mar-14	LIRS Sacrimites Australia Ltd	200			
14-Mar-14	UBS Securities Australia Ltd	Bity	11 352	1 330	Ordinary
14-Mar-14	UBS Securities Australia I td	Bity	305 5	513	ording.
14-Mar-14	UBS Securities Australia I td	Bity	2,200	5	Ordinary
14-Mar-14	LIBC Countries Americal to		1 0,0	1 5	Ordinary 1
14-Mar-14	LIBS Sequities Australia 14d	, a	170	2 2	Ordinary
14-Mar-14	UBS Securities Australia I td	Briv	201	1	Ordinary
14-Mar-14	I IRS Securities Australia I td	to a	1 244	140	Cignal
14-Mar-14	1 BS Securities Australia 1+4	500	3631	2 5	Olumany
14-Mar-14	188 Securities Australia 14d	600	1 755	200	Oldinally
14.Mar.14	TIDE Commission Authority Land	bad.	007,1	2007	Ordinary
	Obs Securities Australia Ltd	gnà	4,320	207	Ordinary
14-IVIA1-14	UBS Securities Australia Ltd	Buy	9,849	1,154	Ordinary
	UBS Securities Australia Ltd	Buy	307	36	Ordinary
- 1	UBS Securities Australia Ltd	Buy	10,737	1,258	Ordinary
- 1	UBS Securities Australia Ltd	Виу	26,048	3,043	Ordinary
- 1	UBS Securities Australia Ltd	Buy	10,576	1,237	Ordinary
- 1	UBS Securities Australia Ltd	Buy	29'6	1,119	Ordinary
	UBS Securities Australia Ltd	Buy	23,882	2,803	Ordinary
	UBS Securities Australia Ltd	Buy	12,276	1,435	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	7,973	932	Ordinary
	UBS Securities Australia Ltd	Buy	22,590	2,639	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	16,343	1.907	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	1,929	225	Ordinary
	UBS Securities Australia Ltd	Buy	34	4	Ordinary
	UBS Securities Australia Ltd	Buy	11,042	1,287	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	43	ı	Ordinary
	UBS Securities Australia Ltd	Buy	17	7	Ordinary
	UBS Securities Australia Ltd	Buy	8,117	946	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	6.843	798	Ordinary
ı	UBS Securities Australia Ltd	Buy	8.575	1,000	Ordinary
ı	UBS Securities Australia Ltd	Buv	197	23	Ordinary
	UBS Securities Australia Ltd	Buy	17.416	2.031	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	6,903	805	Ordinary
ı	UBS Securities Australia Ltd	Buv	8.575	1 000	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buv	10.179	1 187	Ordinac
14-Mar-14	UBS Securities Australia I td	× 8	2 358	375	Ording
	UBS Securities Australia Ltd	Buy	277.2	0001	Called C
L	I IRS Securities Australia 11d	Buck	230.0	1000	Gulliary
1	LIBS Securities Australia I to	Barr	390	5	Ordinary
1	I IBC Constitute Australia 144	600	2002	5	Cidillary
14-Mar-14	LIBS Sequities Australia 14d	8.8	100	000	Gelliary
14-Mar-14	IDO Convision Australia Ltd	in a	106	COL	Ordinary
14 Mar 14	195 Committee Australia Ltd	and a	3,175	3/0	Ordinary
14-IVIAI-14	Ubs securities Australia Ltd	Buy	051,1	134	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	129	15	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	154	18	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	6	-	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	43	2	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	17	2	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	1,201	140	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	8,125	947	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	7.838	913	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	3.752	437	Ordinary
14-Mar-14	UBS Securities Australia Ltd	BC	4.979	580	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	3 631	423	Ordinary
14-Mar-14	UBS Securities Australia I1d	Biv	2048	885	Ordinary
4-Mar-14	UBS Securities Australia Itd	Bity	10.742	1 252	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	2.927	341	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	6.868	800	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Riiv	146	12	Ordinary
4-Mar-14	I IRS Societies Australia 14d	8.00	0000	/ 000	Ordinary
4 Mar 14	Dis Securities Australia Ltd	fna	7,400	507	Orginary
4-Mdr-14	USS SECURITIES AUSTRAINA LTG	Buy	94		Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	10,763	1,253	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	464	54	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Sell	4,046	(471)	Ordinary
				,	
16-Mar-14	UBS Securities Australia Ltd	Sell	3.535	(411)	Ordinary

1000	מכן אבמורו ואמן ומלביו וביור עכים מוום דום	, and	004'/	6/0	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Sin	6 950	010	Creation of
4-Mar-14	I IRC Woolth Management Australia 14d	in a	0,930	1 466	Ordinary
1 May 1	LIDS Woolth Management Australia Ltd	in a	266,21	400	Ordinary
4 14 14	TOO WESTER INTERIOR AUSTRALIA TO	ana a	13,740	800'I	Ordinary
t - IBIAI	ODS Wealth Ividi agement Australia Etd	prix	508,11	388	Ordinary
4-Mar-14	UBS Wealth Mahagement Australia Ltd	Buy	14,549	1,702	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	71,276	8,338	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	23,174	2,711	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	14,626	1,711	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	8/0'6	1,062	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	23,602	2,761	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	16,626	1,945	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	9,343	1,093	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	6.240	730	Ordinary
Mar-14	UBS Wealth Management Australia Ltd	Buv	9 369	1 096	Ordinary
14-Mar-14	LIBS Wealth Management Australia Ltd	Birv	11 771	7751	Ordings.
A-Mar-14	1 JOS Woodth Management Australia 144	600	000,00	1,2,1	Guillaly
1	1 100 Woodsh Management Australia 144	Social Control of the	20,730	57477	Ordinary
4-IVIGIT-14	Obs wealth ivial age then thought Ltd	ens)	945,11	1,35	Ordinary
Mai - 14	Upp weakin Management Australia Ltd	Buy	997'9	/33	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	17,934	2,098	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	5,548	649	Ordinary
Mar-14	UBS Wealth Management Australia Ltd	Buy	27,406	3,206	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Виу	10,660	1,247	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	7,249	848	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	12.746	1.491	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	5.522	646	Ordinary
Mar-14	UBS Wealth Management Australia Ltd	Buy	18 353	2 147	Ordinary
4-Mar-14	I IBC Wealth Management Australia 14d	Busy	970.00	2 475	Ordinary Ordinary
	COST TATELLA MATERIAL PROPERTY AND	, and	0/7:67	5,423	Oldfrary
# ! . I !	ODS Wealth Management Australia Ltd	, and	8//8	020,1	Ordinary
4-Mar-14	UBS wealth Management Australia Ltd	Buy	6,394	748	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	903,213	105,660	Ordinary
4-Mar-14	UBS Wealth Management Australia Ltd	Buy	19,131	2,238	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Sell	536,291	(62,708)	Ordinary
Mar-14	UBS Securities Australia Ltd	AGE A	23 957	2,802	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Birv	23 957	2,802	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Bix	60.055	7.024	Ordinan
A-Mar-14	I IBC Counties Australia Ltd	6	010,00	1,027	Oldinary
Man 14	Opposed the American Line	Buy	050,12	7,402	Orginary
4-Mar-14	UBS Securities Australia Ltd	Buy	132,106	15,451	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	127,096	14,865	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	23,547	2,754	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	34,799	4,070	Ordinary
4-Mar-14	'UBS Securities Australia Ltd	Buy	96,036	206	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buv	18.348	2 146	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	31 370	3 669	Ordinary
Mar-14	I JBS Securities Australia I to	Riv	123 087	14 501	or district
	Data Security Action 2000	(ng	123,304	100,41	Cigilary :
Mar-14	UDS Securities Australia Ltd	Buy	12,455	1,458	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	65,365	7,645	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	17,100	2,000	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	191,734	22,425	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	33,396	3,906	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	308,783	36,115	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	11,619	1,359	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	10.508	1,229	Ordinary
Mar-14	UBS Securities Australia Itd	Bix	43		Ordinan
12.Mar-14	I IRS Securities Australia I to	200	368	1	Ordings Ordings
Mar.14	1 IDC Cooperation Australia 14d	100	000	n (Oldinary
Mai	UBS Securities Australia Liu	Buy	539	£9	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	1,086	127	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	2,112	247	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Виу	1,257	147	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	1,077	126	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Buy	1 838	215	Ordinary
4-Mar-14	I IBS Securities Australia I to	Blay	0.007	203	Ordinary
	COO Securities Australia Etc.	and .	700,0	979	Orginary
4-Mar-14	UBS Securities Australia Ltd	Виу	1,582	185	Ordinary
4-Mar-14	UBS Securities Australia Ltd				-
	The second secon		1115/	6	Ordinary
14-Mar-14	UBS Securities Australia Ltd	Bcv	14.013	1.639	Ordinary

San	8	S S		8	9 4	3 8	SE SE	Š	SE SE	CBS	Sec	8 8	ĕ	3 2	3 8	E S	3 ×	3 8	88	NBS	NBS	UBS	UBS	UBS	SS BS	San	ŝ	BS	9	Sen	Se	nBS	SS PS	8	SS	RBS	nBS	88	Sal	g g	g k	S S	nBS	SE BS	UBS /	'SBO	NBS:	UBS :	San	UBS	San	San Va	Sec	UBS	LBS.	UBS	UBS 5	UBS S	UBS S	Sen	OBS 3	nBS 3	æ
17-Mar-14	17-Mar-14	17-Mar-14	1 /-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	1 /-Mar-14	1 /-Mar-14	7-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	18-Mar-14	18-Mar-14	18-Mar-14	18-Mar-14	18-Mar-14	18-Mar-14	18-Mar-14	10-Mar 14	8-Mar-14	8-Mar-14	8-Mar-14	18-Mar-14	18-Mar-14	18-Mar-14	8-Mar-14	8-Mar-14	- 1	ŀ	18-Mar-14
_													•		•			-			_	1								_		-1									1.							<u>- 1</u>			_1.	-1-	-1-	1-		<u></u>		-1	<u>-1</u>		-1.	=1-	=)
																																			•				_																								
λ	2	2	2	≥	<u> </u>	2	>	2	2	. ≥	2	2		T	. 2	\		\		>	5	2	2	<u>-</u>	2	۸			<u>~</u>	2	2	_	2	2	_	>	2		,	,	, ,	_>	7	Ţ_	_	ý	_	_			Ţ,],	Ţ	_		>	_	À	_		T	Ţ	_
Ordinary	Ordinary	Calp	Orginary	Ordinary	Ordinary	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordinar	Ordinary	Ordina	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordina	Ordinary	Crdina	Ordinary	Ordinary	Ordinary	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinar	Ordinary	Ordinary	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordinary	Ordinar	Ordinary
(487)	(9/4)	3	(1)	(8/4)	(15)	2,000	(1,240,404)	(34,000)	(20,000)	4,567	(5,000)	93,345	(67.665)	(527)	1,315	(209)	(289)	6	(81)	(440)	(448)	(41)	(452)	(537)	(402)	(1/3)	(575)	40	(150)	(57)	(3,350)	(7,7977)	(14,602)	(2,323)	(5,439)	(20)	(77)	(000)	(306)	(1 223)	(350)	(282)	(216)	(88)	(296)	(15)	(65)	(236)	(115)	(401)	(50)	(510)	(368)	(452)	(267)	(138)	(320)	(375)	(287)	(11)	(234)	(75c)	(727)
4,174	4,902	n 0	4010	5 6	128	₹A			ΑN	92	02	8	252	15	22	8	88	77	26	3,780	44	351	3,871	02	33		34	47	5, 3	8 :	10	2 2	33	80	71 8	429	61	813	1 6	33.65	16	21	82	757	56	129	260	15	166	5/4	2 2	2 5	53	37	96	37	25	52	80 1	35	7 5	7,5	×
4,1	2,4		0	4,915	,		_		_	39,2	42,5	802,300	581.4	4.5	11,5	1.7	2,488		٩	3,7	3,8	3	3,8	4,6	3,4	4	4,9	֓֟֟֝֟֓֓֓֓֓֓֓֓֓֓֟֓֓֓֟֓֓֓֟֓֓֓֓֟֓֓֓֓֓֟֓֓֓֟	5,5/3	4 6	28,710	5,22	125,1	19,908	46,612	4 1	1 326	E, C	2,2	10.4	3,007	2,4	6,158	7	5,126		15	4,615	50 0	5,9/4		7. 4	3,423	3,887	2,296	1,1	2,752	3,225	2,468	"	2,012	1,892	7,0
		1	-		\downarrow		L										L													_								1											1				L							\downarrow	\downarrow	\downarrow	
						Stock borrowed	turned	turned	turned																							l			ŀ																												
<u>s</u>	B 3	ž 3	ļ 3	ll de	3	Stock b	Stock returned	Stock returned	Stock returned	Buy	Seil	Buy	Sell	Sell	Buy	, E	- - -	- -	Se.	Sell	Sell	Sell	les Re	Sell	Je J	ě,	ē,	anà	R 3	- E	- S	ē :	la S	a y 3	8 3	9	B 3	= 3	<u> </u>	- N	Sell	Sell	Sell	Sell	Sell	Sel.	J.	3	я 3	Ņ J	3	و	Sell	Sell	Sell	Sell	Se!	Je.	es :	le s	# J	N 3	nac
															alia Ltd																																																
alia Ltd	alla Ltd	alia Ltd	alia Ltd	alia ttd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	anagement	JBS Wealth Management Australia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	əlia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alla Ltd	alla Ltd	alla Lto	in Ltd	alia Lto	alla Ltd	lia Ltu	alla Ltd	alia Ltd	illa Ltd	alia Ltd	alia Ctd	lio Ltd	lia Itd	lia Ltd	lia Ltd	lia Ltd	ilia Ltd	ilia Ltd	ilia Ltd	lia Ltd	lla Ltd	lla Ltd	lia Ltd	lia I to	lia I to	lia Ltd	lia Ltď	ila Ltd	lia Ltd	lia Ltd	lia Ltd	lia Ltd	lia Utd	lia Ltd	lla Lto	a Lta	וא רוח
UBS Securities Australia Ltd	urities Austr	ritios Austr	IBS Securities Australia Ltd	Securities Australia Ltd	urities Austr	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	urities Austr	urities Austr	UBS Securities Australia Ltd	urities Austr	Global Asset Management	alth Manage	urities Austr	UBS Securities Australia Ltd	UBS Securities Australia Ltd	rities Austra	UBS Securities Australia Ltd	UBS Securities Australia Ltd	urities Austr	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Irities Austr	UBS Securities Australia Ltd	UBS Securities Australia Ltd	ITTIES AUSTR	UBS Securities Australia Ltd	irities Austra	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Intles Austra	UBS Securities Australia Ltd	UBS Securities Australia Ltd	irities Austr	UBS Securities Australia (1d	rition Augustra	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	rities Austra	UBS Securities Australia Ltd	DS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JRS Securities Australia Ltd	rities Austra	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	rities Austra	Securities Australia Ltd	Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	rities Austra	UBS Securities Australia Ltd	TUES Ausua			
UBS Sec	28. Ceo.	I BS Se	I By Sey	I BS Sec	1	┮	┢	_	UBS Sea	UBS Sea	UBS Sea	UBS Sect	UBS Sea	UBS Glo	UBS We.	UBS Sea	UBS Sea	UBS Sea	UBS Sec	UBS Sea	UBS Sea	UBS Sea	UBS Sea	UBS Sea	UBS Sed	Des sed	UBS Sect	Sec con	Dec con	Dec Sec	Des sed	Dac con	UBS Sect	Des sed	Day Sec	OBS SEC	UBS Sea	I BS Sed	UBS Sea	UBS Sec.	UBS Sea	UBS Sect	UBS Sea	UBS Sect	UBS Sect	UBS Sect	UBS Sect	UBS Sea	Des Sed	UBS Sect	1 IRS Section	UBS Secu	UBS Secu	UBS Secu	UBS Secu	UBS Sea	UBS Secu	UBS Secu	UBS Sea	UBS Sec.	I BS Sec.	UBS Secu	Upp Seve
6-Mar-14	6-Mar-14	16-Mar.14	16-Mar-14	16-Mar-14	16-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	1 /-Mar-14	1 /-Mar-14	17 Mar-14	17 - Widi - 14	17-Mar 14	7 - INIAI - 14	7-Mar-14	17 May 14	17 Mar 14	7 Mar 14	17 Mar 14	7 Mar-14	7-Mar-14	7-Mar-14	7-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	17-Mar-14	7-Mar-14	/-Mar-14	7 Mar-14	7-Mar-14	7-Mar-14	7-Mar-14	7-Mar-14	17-Mar-14	7-Mar-14	7-Mar-14	7-Mar-14	7-Mar-14	7-Mar-14	7-Mar-14	17-Mar-14	7-Mar-14	17-Mar-14	/-lv:a;-++

AMERIA I. 80. Scorning Audining LM Sell 1,58 (16) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell Sell (18) (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell Sell (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell Sell (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell (10) Orderay 1/20 (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell (10) Orderay 1/20 (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell (10) Orderay 1/20 (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell (10) Orderay 1/20 (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell (10) Orderay 1/20 (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell (10) Orderay 1/20 (10) Orderay 1/20 AMERIA I. 80. Scorning Audining LM Sell (10) Orderay 1/20 (10) Ord						
USS Securities Australia Ltd Sell Sell	-Mar 14	UBS Securities Australia Ltd	- Se	1,385	(161)	Ordinary
USS Securities Australia Ltd Sell Sell	-Mar-14	LIRS Securities Australia Ind	Jac.	98 55	(0)	Ordinary
USS Securities Australia Ltd Sell Sell	Mar	A IDS Counties Australia Ltd	136	\$ 5	(61)	Ordinary
USS Securities Australia Ltd Sell Sell	17 - IPINI-	Cops securities Australia Ltd) Aei	90	(2)	Ordinary
USS Securities Australia Ltd Sell 2,947 2727 2728 2729 272	- Inlai	Coo securities Australia Ltd	Sell	98	(01)	Ordinary
URS Securities Australia Ltd Sell 2,342 27.27 27.28	t - INIGI.	Obs Securities Australia Ltd	Sell	90	(2)	Ordinary
UBS Securities Australia Ltd Sell 2,342 (27) UBS Securities Australia Ltd Sell 2,391 (27) UBS Securities Australia Ltd Sell 2,391 (27) UBS Securities Australia Ltd Sell 2,391 (27) UBS Securities Australia Ltd Sell 1,514 (17) UBS Securities Australia Ltd Sell 1,524 (17) UBS Securities Australia Ltd Sell 2,242 (27) UBS Securities Australia Ltd Sell 3,242 (27) UBS Securities Australia Ltd Sell 3,242 (27) UBS Securities Australia Ltd <	-Mar-14	UBS Securities Australia Ltd	Seli	6	(1)	Ordinary
UBS Securities Australia Ltd Sell 2.95 (270) UBS Securities Australia Ltd Sell 2.75 (270) UBS Securities Australia Ltd Sell 2.95 (87) UBS Securities Australia Ltd Sell 2.96 (87) UBS Securities Australia Ltd Sell 2.546 (72) UBS Securities Australia Ltd Sell 2.546 (72) UBS Securities Australia Ltd Sell 1.61 (17) UBS Securities Australia Ltd Sell 1.63 (17) UBS Securities Australia Ltd Sell 1.63 (17) UBS Securities Australia Ltd Sell 1.63 (17) UBS Securities Australia Ltd Sell 1.64 (17) UBS Securities Australia Ltd Sell 3.28 (17) UBS Securities Australia Ltd Sell 3.26 (21) UBS Securities Australia Ltd Sell 3.61 4.35 (4.42) UBS Securities Australia Ltd Sell 3.61 1.67 (17) UBS Secu	-Mar-14	UBS Securities Australia Ltd	Sell	2,342	(272)	Ordinary
UBS Securities Australia Ltd Seal 2.391 (278) LDBS Securities Australia Ltd Seal 2.391 (278) LDBS Securities Australia Ltd Seal 2.546 (787) LDBS Securities Australia Ltd Seal 1.514 (178) LDBS Securities Australia Ltd Seal 1.514 (178) LDBS Securities Australia Ltd Seal 1.514 (178) LDBS Securities Australia Ltd Seal 2.546 (179) LDBS Securities Australia Ltd Seal 2.546 (170) LDBS Securities Australia Ltd Seal 2.546 (170) LDBS Securities Australia Ltd Seal 1.637 (171) LDBS Securities Australia Ltd Seal 1.643 (171) LDBS Securities Australia Ltd Seal 1.643 (172) LDBS Securities Australia Ltd Seal 2.765 (173) LDBS Securities Australia Ltd Seal 2.765 (173) LDBS Securities Australia Ltd Seal 2.765 (173) LDBS Securitie	-Mar-14	UBS Securities Australia Ltd	Sell	17	(2)	Ordinary
UBS Securities Australia Ltd Sell Sell 695 617 UBS Securities Australia Ltd Sell Sell 619 619 UBS Securities Australia Ltd Sell Sell 1541 1540 (472) UBS Securities Australia Ltd Sell Sell 1541 1540 (472) UBS Securities Australia Ltd Sell Sell 1540 (472) UBS Securities Australia Ltd Sell Sell 1688 (103) UBS Securities Australia Ltd Sell Sell 3,810 (442) (13) UBS Securities Australia Ltd Sell Sell 3,810 (442) (13) UBS Securities Australia Ltd Sell Sell 3,810 (13) (13) UBS Securities Australia Ltd Sell Sell 3,810 (13) (13) (13) UBS Securities Australia Ltd Sell Sell 3,810 (13)	-Mar-14	UBS Securities Australia Ltd	Sell	2,391	(278)	Ordinary
USS Securities Australia Ltd Seil Seil Seil Seil Seil Seil Securities Australia Ltd Seil Securities Australia Ltd Seil Securities Australia Ltd Seil	-Mar-14	UBS Securities Australia Ltd	Sell	275	(32)	Ordinary
UBS Securities Australia Ltd Sell 1,544 (170) 105 Securities Australia Ltd Sell 1,644 (170) 1,645	7-Mar-14	UBS Securities Australia Ltd	Seil	69	(8)	Ordinary
UBS Securities Australia Ltd Sell 1,554 (178) UBS Securities Australia Ltd Sell 2,566 (27) UBS Securities Australia Ltd Sell 2,566 (27) UBS Securities Australia Ltd Sell 1,558 (216) UBS Securities Australia Ltd Sell 1,658 (216) UBS Securities Australia Ltd Sell 1,688 (171) UBS Securities Australia Ltd Sell 1,678 (171) UBS Securities Australia Ltd Sell 1,679 (173) UBS Securities Australia Ltd Sell 1,670 (173) UBS Securities Australia Ltd Sell 2,675 (173) UBS Securities Australia Ltd Sell 2,675 (174) UBS Securities Australia Ltd </td <td>7-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>lias</td> <td>U26 E</td> <td>(757)</td> <td>Ordinan</td>	7-Mar-14	UBS Securities Australia Ltd	lias	U26 E	(757)	Ordinan
UBS Securities Australia Ltd Sell 4,059 (472) UBS Securities Australia Ltd Sell 1,546 (472) UBS Securities Australia Ltd Sell 1,886 (103) UBS Securities Australia Ltd Sell 1,896 (710) UBS Securities Australia Ltd Sell 1,896 (711) UBS Securities Australia Ltd Sell 3,286 (812) UBS Securities Australia Ltd Sell 3,286 (817) UBS Securities Australia Ltd Sell 4,787 (73) UBS Securities Australia Ltd Sell 3,267 (82) UBS Securities Australia Ltd Sell 4,787 (74) UBS Securities Australia Ltd <td>7-Mar-14</td> <td>UBS Sequities Australia I td</td> <td>li S</td> <td>1 514</td> <td>(321)</td> <td>Ordinary</td>	7-Mar-14	UBS Sequities Australia I td	li S	1 514	(321)	Ordinary
URS Securities Australia Ltd Sell 2,568 (47) URS Securities Australia Ltd Sell 1,815 (71) URS Securities Australia Ltd Sell 1,826 (71) URS Securities Australia Ltd Sell 1,826 (71) URS Securities Australia Ltd Sell 1,873 (91) URS Securities Australia Ltd Sell 1,875 (71) URS Securities Australia Ltd Sell 2,675 (72) URS Securities Australia Ltd Sell 2,675 (73) URS Securities Australia Ltd Sell 2,675 (74) URS Securities Australia Ltd Sell 2,675 (74) URS Securities Australia Ltd <	7-Mar-14	I IBS Secretials Australia 14d	lleg Ileg	#16,1	(0/1)	Ordinary
UBS Securities Australia Ltd Sell 1,590 (1,590) UBS Securities Australia Ltd Sell 1,866 (1,01) UBS Securities Australia Ltd Sell 1,866 (1,01) UBS Securities Australia Ltd Sell 1,643 (1,01) UBS Securities Australia Ltd Sell 4,555 (1,01) UBS Securities Australia Ltd Sell 4,555 (1,01) UBS Securities Australia Ltd Sell 4,555 (1,01) UBS Securities Australia Ltd Sell 2,766 (1,02) UBS Securities Australia Ltd Sell 2,766 (1,02) UBS Securities Australia Ltd Sell 2,766 (1,02) UBS Securities Australia Ltd Sell 3,771 (1,02) UBS Securiti	Mar 14	LIDS Counties Australia Ltd	llac I	4,059	(4/2)	Ordinary
UBS Securities Australia Ltd Sell 185 (2.1) UBS Securities Australia Ltd Sell 1,856 (2.1) UBS Securities Australia Ltd Sell 1,866 (2.1) UBS Securities Australia Ltd Sell 1,875 (2.1) UBS Securities Australia Ltd Sell 1,273 (1.2) UBS Securities Australia Ltd Sell 1,475 (2.1) UBS Securities Australia Ltd Sell 1,475 (2.1) UBS Securities Australia Ltd Sell 1,47 (1.2) UBS Securities Australia Ltd Sell 1,47 (1.2) UBS Securities Australia Ltd Sell 2,57 (1.2) UBS Securities Australia Ltd Sell 2,67 (1.2) UBS Securities Australia Ltd Sell 3,50 (1.2) UBS Securities Australia Ltd	-IVIGI - 14	Obsideration Australia Ltd	Sell	2,546	(296)	Ordinary
UBS Securities Australia Ltd Sell 1886 (103) UBS Securities Australia Ltd Sell 188 (216) UBS Securities Australia Ltd Sell 188 (216) UBS Securities Australia Ltd Sell 1,043 (217) UBS Securities Australia Ltd Sell 1,043 (217) UBS Securities Australia Ltd Sell 3,208 (387) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 4,333 (505) UBS Securities Australia Ltd Sell 4,333 (4,13) UBS Securities Australia Ltd <td>-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Self</td> <td>181</td> <td>(21)</td> <td>Ordinary</td>	-Mar-14	UBS Securities Australia Ltd	Self	181	(21)	Ordinary
UBS Securities Australia Ltd Sell 1.888 (21) UBS Securities Australia Ltd Sell 1.888 (21) UBS Securities Australia Ltd Sell 284 (33) UBS Securities Australia Ltd Sell 7.83 (17) UBS Securities Australia Ltd Sell 7.87 (17) UBS Securities Australia Ltd Sell 7.97 (17) UBS Securities Australia Ltd Sell 7.07 (82) UBS Securities Australia Ltd Sell 3.80 (3.00) UBS Securities Australia Ltd Sell 3.76 (3.00) UBS Securities Australia Ltd Sell 3.70 (4.77) UBS Securities Australia Ltd Sell<	-Mar-14	UBS Securities Australia Ltd	Sell	886	(103)	Ordinary
UBS Securities Australia Ltd Sell 181 (21) UBS Securities Australia Ltd Sell 28 (33) UBS Securities Australia Ltd Sell 1,043 (121) UBS Securities Australia Ltd Sell 1,043 (172) UBS Securities Australia Ltd Sell 1,285 (185) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 3,301 (471) UBS Securities Australia Ltd Sell 2,65 (171) UBS Securities Australia Ltd Sell 3,801 (4,42) UBS Securities Australia Ltd <td>-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>1,858</td> <td>(216)</td> <td>Ordinary</td>	-Mar-14	UBS Securities Australia Ltd	Sell	1,858	(216)	Ordinary
UBS Securities Australia Ltd Sell 788 (37) UBS Securities Australia Ltd Sell 1,043 (121) UBS Securities Australia Ltd Sell 3,203 (172) UBS Securities Australia Ltd Sell 3,203 (172) UBS Securities Australia Ltd Sell 1,47 (173) UBS Securities Australia Ltd Sell 1,47 (173) UBS Securities Australia Ltd Sell 1,47 (173) UBS Securities Australia Ltd Sell 1,77 (15) UBS Securities Australia Ltd Sell 1,77 (174) UBS Securities Australia Ltd Sell 2,675 (505) UBS Securities Australia Ltd Sell 3,811 (4,77) UBS Securities Australia Ltd Sell 3,817 (4,77) UBS Securities Australia Ltd <td>-Mar-14</td> <td>UBS securities Australia Ltd</td> <td>Sell</td> <td>181</td> <td>(21)</td> <td>Ordinary</td>	-Mar-14	UBS securities Australia Ltd	Sell	181	(21)	Ordinary
UBS Securities Australia Ltd Sell 1783 (91) UBS Securities Australia Ltd Sell 1,023 (171) UBS Securities Australia Ltd Sell 3,203 (372) UBS Securities Australia Ltd Sell 1,875 (277) UBS Securities Australia Ltd Sell 1,77 (28) UBS Securities Australia Ltd Sell 1,70 (42) UBS Securities Australia Ltd Sell 2,65 (15) UBS Securities Australia Ltd Sell 2,65 (173) UBS Securities Australia Ltd Sell 2,65 (173) UBS Securities Australia Ltd Sell 3,71 (173) UBS Securities Australia Ltd Sell 3,71 (173) UBS Securities Australia Ltd Sell 3,71 (174) UBS Securities Australia Ltd	-Mar-14	UBS Securities Australia Ltd	Sell	284	(33)	Ordinary
UBS Securities Australia Ltd Sell 1,043 (121) UBS Securities Australia Ltd Sell 3,298 (372) UBS Securities Australia Ltd Sell 3,298 (373) UBS Securities Australia Ltd Sell 4,186 (482) UBS Securities Australia Ltd Sell 7,24 (28) UBS Securities Australia Ltd Sell 7,27 (17) UBS Securities Australia Ltd Sell 7,27 (17) UBS Securities Australia Ltd Sell 7,27 (17) UBS Securities Australia Ltd Sell 4,558 (37) UBS Securities Australia Ltd Sell 4,558 (37) UBS Securities Australia Ltd Sell 4,793 (47) UBS Securities Australia Ltd Sell 2,675 (52) UBS Securities Australia Ltd Sell 4,793 (47) UBS Securities Australia Ltd Sell 3,595 (4,18) UBS Securities Australia Ltd Sell 3,595 (4,18) UBS Securities Australia Ltd	'-Mar-14	UBS Securities Australia Ltd	Sell	783	(16)	Ordinary
UBS Securities Australia Ltd Sell 3,203 (372) UBS Securities Australia Ltd Sell 1,286 (485) UBS Securities Australia Ltd Sell 1,47 (77) UBS Securities Australia Ltd Sell 1,47 (17) UBS Securities Australia Ltd Sell 1,47 (17) UBS Securities Australia Ltd Sell 1,70 (42) UBS Securities Australia Ltd Sell 2,769 (170) UBS Securities Australia Ltd Sell 3,171 (47) UBS Securities Australia Ltd Se	'-Mar-14	UBS Securities Australia Ltd	Sell	1,043	(121)	Ordinary
UBS Securities Australia Ltd Sell 1,875 (377) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 1,47 (17) UBS Securities Australia Ltd Sell 1,77 (52) UBS Securities Australia Ltd Sell 1,77 (52) UBS Securities Australia Ltd Sell 3,801 (4,422) UBS Securities Australia Ltd Sell 3,801 (4,422) UBS Securities Australia Ltd Sell 2,675 (131) UBS Securities Australia Ltd Sell 2,675 (132) UBS Securities Australia Ltd Sell 3,595 (1,620) UBS Securities Austr	7-Mar-14	UBS Securities Australia Ltd	Sell	3,203	(372)	Ordinary
UBS Securities Australia Ltd Sell 4,186 (217) UBS Securities Australia Ltd Sell 4,186 (485) UBS Securities Australia Ltd Sell 147 (17) UBS Securities Australia Ltd Sell 707 (82) UBS Securities Australia Ltd Sell 3,801 (442) UBS Securities Australia Ltd Sell 3,801 (442) UBS Securities Australia Ltd Sell 3,801 (442) UBS Securities Australia Ltd Sell 2,655 (31) UBS Securities Australia Ltd Sell 3,507 (477) UBS Securities Australia Ltd Sell 3,550 (478) UBS Securities Australia Burch Buy 2,560 (100) UBS Securities Australia Ltd Sell 3,570 (4,78) UBS Securities Australia Ltd	7-Mar-14	UBS Securities Australia Ltd	les	3,288	(381)	Ordinary
UBS Securities Australia Ltd Sell 4.186 (485) UBS Securities Australia Ltd Sell 1.29 (15) UBS Securities Australia Ltd Sell 1.29 (15) UBS Securities Australia Ltd Sell 4,353 (505) UBS Securities Australia Ltd Sell 4,353 (505) UBS Securities Australia Ltd Sell 4,353 (505) UBS Securities Australia Ltd Sell 4,555 (507) UBS Securities Australia Ltd Sell 2,675 (107) UBS Securities Australia Ltd Sell 2,765 (107) UBS Securities Australia Ltd Sell 3,675 (107) UBS Securities Australia Ltd Sell 3,675 (477) UBS Securities Australia Ltd Sell 3,171 (479) UBS Securities Australia Ltd <td>7-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Seli</td> <td>1,875</td> <td>(217)</td> <td>Ordinary</td>	7-Mar-14	UBS Securities Australia Ltd	Seli	1,875	(217)	Ordinary
UBS Securities Australia Ltd Sell 242 (12) UBS Securities Australia Ltd Sell 147 (17) UBS Securities Australia Ltd Sell 707 (82) UBS Securities Australia Ltd Sell 4,333 (74) UBS Securities Australia Ltd Sell 4,333 (805) UBS Securities Australia Ltd Sell 4,333 (805) UBS Securities Australia Ltd Sell 4,333 (805) UBS Securities Australia Ltd Sell 4,375 (81) UBS Securities Australia Ltd Sell 8,507 (41) UBS Securities Australia Ltd Sell 2,675 (81) UBS Securities Australia Ltd Sell 3,77 (41) UBS Securities Australia Ltd Buy </td <td>'-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Seli</td> <td>4,186</td> <td>(482)</td> <td>Ordinary</td>	'-Mar-14	UBS Securities Australia Ltd	Seli	4,186	(482)	Ordinary
UBS Securities Australia Ltd Sell 147 (17) UBS Securities Australia Ltd Sell 707 (82) UBS Securities Australia Ltd Sell 707 (82) UBS Securities Australia Ltd Sell 4,353 (505) UBS Securities Australia Ltd Sell 3,801 (442) UBS Securities Australia Ltd Sell 2,675 (31) UBS Securities Australia Ltd Sell 3,597 (477) UBS Securities Australia Ltd Sell 3,590 (477) UBS Securities Australia Ltd Buy 2,530 (478) UBS Securities Australia Ltd	7-Mar-14	UBS Securities Australia Ltd	Sell	242	(28)	Ordinary
UBS Securities Australia Ltd Sell (15) UBS Securities Australia Ltd Sell (707 (82) UBS Securities Australia Ltd Sell 638 (704) UBS Securities Australia Ltd Sell 4,553 (505) UBS Securities Australia Ltd Sell 3,801 (442) UBS Securities Australia Ltd Sell 3,601 (472) UBS Securities Australia Ltd Sell 2,675 (513) UBS Securities Australia Ltd Sell 2,675 (418) UBS Securities Australia Ltd Sell 3,595 (418) UBS Securities Australia Ltd Sell 3,271 (4,197) UBS Securities Australia Ltd Buy 2,560 (5,000) UBS Securities Australia Ltd Buy <td>7-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>147</td> <td>(71)</td> <td>Ordinary</td>	7-Mar-14	UBS Securities Australia Ltd	Sell	147	(71)	Ordinary
UBS Securities Australia Ltd Sell 707 (82) UBS Securities Australia Ltd Sell 4,353 (74) UBS Securities Australia Ltd Sell 4,353 (605) UBS Securities Australia Ltd Sell 4,565 (522) UBS Securities Australia Ltd Sell 4,565 (532) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 2,669 (100) UBS Securities Australia Ltd Sell 2,769 (477) UBS Securities Australia Ltd Sell 2,769 (477) UBS Securities Australia Ltd Sell 3,77 (478) UBS Securities Australia Ltd Sell 3,77 (478) UBS Securities Australia Ltd Sell 3,595 (4,18) UBS Securities Australia Ltd Sell 3,520 (4,18) UBS Securities Australia Ltd Buy 4,210 (4,27) UBS Securities Australia Ltd Buy 4,210 (4,07) UBS Securities Australia Ltd <td>7-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>129</td> <td>(15)</td> <td>Ordinary</td>	7-Mar-14	UBS Securities Australia Ltd	Sell	129	(15)	Ordinary
UBS Securities Australia Ltd Sell 4,353 (74) UBS Securities Australia Ltd Sell 4,353 (505) UBS Securities Australia Ltd Sell 1,7 (7) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 2,695 (477) UBS Securities Australia Ltd Sell 3,595 (477) UBS Securities Australia Ltd Buy 3,597 (477) UBS Securities Australia Ltd Buy 3,597 (477) UBS Securities Australia Ltd Buy 3,597 (478) UBS Securities Australia Ltd	'-Mar-14	UBS Securities Australia I td	0)	202	(5.8)	Ordinary,
UBS Securities Australia Ltd Sell 4,353 (505) UBS Securities Australia Ltd Sell 4,353 (505) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 2,675 (322) UBS Securities Australia Ltd Sell 2,675 (418) UBS Securities Australia Ltd Sell 3,771 (477) UBS Securities Australia Ltd Sell 3,771 (418) UBS Securities Australia Ltd Sell 3,771 (418) UBS Securities Australia Ltd Buy 2,530,09 2,543 UBS Securities Australia Ltd Buy 2,530,09 2,543 UBS Securities Australia Ltd Buy 2,530,09 2,543 UBS Securities Australia Ltd Buy 3,520 2,543 UBS Securities Australia Ltd Buy 2,569 3,000 UBS Securities Australia	-Mar-14	I IRC Counties Australia 1+4	====	707	(70)	Oldinary
UBS Securities Australia Ltd Sell 4,353 (30) UBS Securities Australia Ltd Sell 3,801 (422) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 2,695 (477) UBS Securities Australia Ltd Sell 3,171 (477) UBS Securities Australia Ltd Sell 3,171 (477) UBS Securities Australia Ltd Sell 3,171 (478) UBS Securities Australia Ltd Sell 3,171 (418) UBS Securities Australia Ltd Sell 3,171 (418) UBS Securities Australia Ltd Sell 3,527 (418) UBS Securities Australia Ltd Buy 35,200 5,000 UBS Securities Australia Ltd Buy 35,200 5,000 UBS Securities Australia Ltd Buy 35,003 1,000 UBS Securities Australia Ltd<	-Mar-14	I IRC Countries Australia I to	llac.	650	(74)	Orginary
UBS Securities Australia Ltd Sell 3,801 (42) UBS Securities Australia Ltd Sell 3,601 (442) UBS Securities Australia Ltd Sell 2,675 (311) UBS Securities Australia Ltd Sell 860 (100) UBS Securities Australia Ltd Sell 2,769 (322) UBS Securities Australia Ltd Sell 4,797 (427) UBS Securities Australia Ltd Sell 3,595 (418) UBS Securities Australia Ltd Sell 3,171 (477) UBS Securities Australia Ltd Sell 3,171 (477) UBS Securities Australia Ltd Buy 1,084,300 1,0,671 UBS Securities Australia Ltd Buy 1,0,64,100 5,0,00 UBS Securities Australia Ltd Buy 2,5,000 1,0,00 UBS Securitie	May 1.4	Coop Securities Australia Ltd	E C	4,353	(cnc)	Orainary
UBS Securities Australia Ltd Sell 4.567 (42) UBS Securities Australia Ltd Sell 4.567 (311) UBS Securities Australia Ltd Sell 2.675 (322) UBS Securities Australia Ltd Sell 2.769 (100) UBS Securities Australia Ltd Sell 2.769 (477) UBS Securities Australia Ltd Sell 3.771 (478) UBS Securities Australia Ltd Sell 3.731 (477) UBS Securities Australia Ltd Sell 3.771 (4.18) UBS Securities Australia Ltd Sell 3.771 (4.11) UBS Securities Australia Ltd Buy 2.530 (4.15) UBS Securities Australia Ltd Buy 2.530 2.94.3 UBS Securities Australia Ltd Buy 3.25,000 2.95.63 UBS Securities Australia Ltd Buy 4.2.780 5.000 UBS Securities Australia Ltd Buy 2.5680 3.000 UBS Securities Australia Ltd Buy 2.5680 3.000 UBS Securities Austral	-Mdr-14	UBS Securities Australia Ltd	Sell	17	(2)	Ordinary
UBS Securities Australia Ltd Sell 4,565 (522) UBS Securities Australia Ltd Sell 26 (131) UBS Securities Australia Ltd Sell 26 (100) UBS Securities Australia Ltd Sell 2,697 (477) UBS Securities Australia Ltd Sell 3,171 (477) UBS Securities Australia Brach Buy 3,55,009 12,661 UBS Securities Australia Brach Buy 3,53,009 1,000 UBS Securities Australia Brach Buy 3,53,009 1,000 UBS Securities Australia Ltd Buy 3,53,009 1,000 UBS Securitie	Mar-14	UBS Securities Australia Ltd	Sell	3,801	(442)	Ordinary
UBS Securities Australia Ltd Sell 2,675 (31) UBS Securities Australia Ltd Sell 2,6 (31) UBS Securities Australia Ltd Sell 4,793 (322) UBS Securities Australia Ltd Sell 4,793 (477) UBS Securities Australia Ltd Sell 4,793 (477) UBS Securities Australia Ltd Sell 3,595 (418) UBS Securities Australia Ltd Sell 3,717 (32) UBS Securities Australia Branch Buy 25,300 29,541 UBS Securities Australia Branch Buy 25,300 29,541 UBS Securities Australia Ltd Buy 25,300 29,541 UBS Securities Australia Ltd Buy 35,271 4,115 UBS Securities Australia Ltd Buy 35,271 4,115 UBS Securities Australia Ltd Buy 3,280 (5,000) UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australi	-Mar-14	UBS Securities Australia Ltd	Seil	4,565	(532)	Ordinary
UBS Securities Australia Ltd Sell 86 (10) UBS Securities Australia Ltd Sell 2,769 (100) UBS Securities Australia Ltd Sell 4,997 (477) UBS Securities Australia Ltd Sell 4,997 (477) UBS Securities Australia Ltd Sell 3,171 (370) UBS Securities Australia Ltd Sell 3,171 (418) UBS Securities Australia Ltd Sell 3,295 (418) UBS Securities Australia Ltd Sell 3,25 (418) UBS Securities Australia Ltd Buy 25,300 29,53 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd	-Mar-14	UBS Securities Australia Ltd	Sell	2,675	(311)	Ordinary
UBS Securities Australia Ltd Sell 860 (100) UBS Securities Australia Ltd Sell 2,769 (477) UBS Securities Australia Ltd Sell 4,479 (477) UBS Securities Australia Ltd Sell 3,597 (477) UBS Securities Australia Ltd Sell 3,771 (477) UBS Securities Australia Ltd Sell 3,771 (477) UBS Securities Australia Branch Buy 3,750 (478) UBS Securities Australia Branch Buy 3,571 4,18 UBS Securities Australia Branch Buy 3,571 4,115 UBS Securities Australia Branch Buy 3,527 4,115 UBS Securities Australia Ltd Buy 3,527 4,115 UBS Securities Australia Ltd Buy 4,2,800 5,000 UBS Securities Australia Ltd Buy 3,54,48 (4,2,54) UBS Securities Australia Ltd Buy 3,56,48 (4,2,73) UBS Securities Australia Ltd Buy 3,56,48 (4,77) UBS Securitie	-Mar-14	UBS Securities Australia Ltd	Sell	26	(3)	Ordinary
UBS Securities Australia Ltd Sell 2,769 (327) UBS Securities Australia Ltd Sell 4,097 (477) UBS Securities Australia Ltd Sell 3,595 (4.18) UBS Securities Australia Ltd Sell 3,77 (32) UBS Securities Australia Ltd Sell 3,77 (4) UBS Securities Australia Branch Buy 23,500 29,540 UBS Securities Australia Branch Buy 23,500 29,541 UBS Securities Australia Branch Buy 23,271 4,115 UBS Securities Australia Branch Buy 23,271 4,115 UBS Securities Australia Ltd Buy 35,271 4,115 UBS Securities Australia Ltd Buy 35,271 4,115 UBS Securities Australia Ltd Buy 36,060 5,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 2,254 4,273 UBS Securities Aus	-Mar-14	UBS Securities Australia Ltd	Sell	860	(100)	Ordinary
UBS Securities Australia Ltd Sell 4,097 (477) UBS Securities Australia Ltd Sell 3,171 (322) UBS Securities Australia Ltd Sell 3,171 (370) UBS Securities Australia Ltd Sell 3,171 (41) UBS Securities Australia Ltd Sell 3,271 (41) UBS Securities Australia Ltd Buy 1,08,430 1,2,661 UBS Acquarties Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 25,680 3,000 UBS Securities Australia	-Mar-14	UBS Securities Australia Ltd	Sell	2,769	(322)	Ordinary
UBS Securities Australia Ltd Sell 3,479 (522) UBS Securities Australia Ltd Sell 3,595 (4,18) UBS Securities Australia Ltd Sell 3,171 (370) UBS Securities Australia Ltd Buy 3,26 (38) UBS Securities Australia Branch Buy 108,430 29,543 UBS Act Australia Branch Buy 35,271 4,115 UBS Securities Australia Ltd Buy 35,271 4,115 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 25,689 10,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,689 10,000 UBS Securities Australia Ltd Buy 25,689 3,000 UBS Securities Australia Ltd Sell 2,3412 UBS Securities Australia Ltd Sel	-Mar-14	UBS Securities Australia Ltd	Sell	4,097	(477)	Ordinary
UBS Securities Australia Ltd Sell 3,595 (4.18) UBS Securities Australia Ltd Sell 3,171 (370) UBS Securities Australia Ltd Sell 326 (38) UBS Securities Australia Branch Buy 23,300 12,661 UBS Securities Australia Branch Buy 23,300 29,533 UBS Securities Australia Ltd Buy 42,800 29,533 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 36,484 (42,554) UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 2,254 (437) UBS Securities Australia Ltd Sell 2,259 (437) UBS Securities Australia Ltd Sell 2,256 (437) UBS Securities Aus	-Mar-14	UBS Securities Australia Ltd	Seli	4,479	(522)	Ordinary
UBS Securities Australia Ltd Sell 3,171 (370) UBS Securities Australia Ltd Sell 34 (4) UBS Securities Australia Brach Buy 25,3009 12,661 UBS Securities Australia Brach Buy 25,3009 29,543 UBS Securities Australia Brach Buy 35,270 29,543 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Sell 36,484 (42,554) UBS Securities Australia Ltd Buy 25,680 1,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 23,487 (475) UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 23,412 (475) UBS Securitie	7-Mar-14	UBS Securities Australia Ltd	Sell	3,595	(418)	Ordinary
UBS Securities Australia Ltd Sell 34 (4) UBS AG Australia Ltd Sell 326 (38) UBS AG Australia Branch Buy 108,430 12,641 UBS Securities Australia Branch Buy 253,009 29,543 UBS Securities Australia Ltd Buy 4115 UBS Securities Australia Ltd Buy 42,000 5,000 UBS Securities Australia Ltd Buy 43,050 (5,000) UBS Securities Australia Ltd Buy 42,183 16,622 UBS Securities Australia Ltd Buy 86,048 10,000 UBS Securities Australia Ltd Buy 26,636 2,412 UBS Securities Australia Ltd Buy 20,636 2,412 UBS Securities Australia Ltd Sell 20,536 2,412 UBS Securities Australia Ltd Sell 20,536 2,437 UBS Securities Australia Ltd Sell 2,340 (7,737) UBS Securities Australia Ltd Sell 2,340 (7,737) UBS Securities Australia Ltd Sell	7-Mar-14	UBS Securities Australia Ltd	Sell	3,171	(370)	Ordinary
UBS Securities Australia Ltd Sell 326 (38) UBS Activative Branch Buy 10,8430 12,661 UBS Activative Branch Buy 28,3009 29,533 UBS Securities Australia Ltd Buy 42,800 29,533 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 42,183 16,522 UBS Securities Australia Ltd Sell 36,484 (42,554) UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 2,053 (4,273) UBS Securities Australia Ltd Sell 2,29,32 (4,477) UBS Securities Australia Ltd Sell 2,29,32 (4,477) UBS Securities Australia Ltd Sell 2,29 3,01 UBS Securities Australia Ltd Sell 3,01 (4,76) UBS Securities Australia Lt	-Mar-14	UBS Securities Australia Ltd	Sell	34	(4)	Ordinary
UBS ACG Australia Branch Buy 108,430 12,661 UBS ACG Australia Branch Buy 25,300 12,661 UBS Securities Australia Ltd Buy 35,300 5,643 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 86,089 10,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Buy 4,075 (476) UBS Securities Australia	-Mar-14	UBS Securities Australia Ltd	Sell	326	(38)	Ordinary
UBS Securities Australia Enarch Buy 253,009 29,543 UBS Securities Australia Ltd Buy 35,271 4,115 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 142,183 16,622 UBS Securities Australia Ltd Buy 142,183 16,622 UBS Securities Australia Ltd Buy 86,484 42,554 UBS Securities Australia Ltd Buy 86,484 42,554 UBS Securities Australia Ltd Buy 26,680 3,000 UBS Securities Australia Ltd Buy 20,636 2,412 UBS Securities Australia Ltd Sell 20,536 2,412 UBS Securities Australia Ltd Sell 2,23,401 (2,737) UBS Securities Australia Ltd Sell 2,3401 (2,737) UBS Securities Australia Ltd Sell 3,415 (369) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Sell 4,075 (476) UBS Secu	-Mar-14	UBS AG Australia Branch	Buy	108 430	12 661	Ording
UBS Securities Australia Ltd Buy 55,271 4,115 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Buy 142,183 16,620 UBS Securities Australia Ltd Buy 36,689 10,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 20,536 3,417 UBS Securities Australia Ltd Sell 2,25,880 3,417 UBS Securities Australia Ltd Sell 2,25,880 3,417 UBS Securities Australia Ltd Sell 2,29,322 2,412 UBS Securities Australia Ltd Sell 2,29 (476) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Buy 1,656 1,657 UBS Securities Australia Ltd Buy 1,677 188 UBS Securities Austra	-Mar-14	UBS AG Australia Branch	Bun	253,000	20 543	Ordinary
UBS Securities Australia Ltd Buy 42,800 5,000 UBS Securities Australia Ltd Sell 42,800 5,000 UBS Securities Australia Ltd Buy 142,183 16,622 UBS Securities Australia Ltd Buy 86,089 10,000 UBS Securities Australia Ltd Buy 86,089 10,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 20,9322 2,412 UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Buy 1,656 1,656 UBS Securities Australia Ltd Buy 1,657 1,89 UBS Securities Aus	3-Mar-14	UBS Securities Australia 11d	B. B.	35 271	4 115	Ordinary Ordinary
UBS Securities Australia Ltd Sell 4,2500 UBS Securities Australia Ltd Buy 142,183 1,622 UBS Securities Australia Ltd Buy 142,183 1,622 UBS Securities Australia Ltd Buy 86,089 10,000 UBS Securities Australia Ltd Buy 86,089 10,000 UBS Securities Australia Ltd Buy 2,5680 2,412 UBS Securities Australia Ltd Sell 2,000 2,003 UBS Securities Australia Ltd Sell 2,3401 (2,737) UBS Securities Australia Ltd Sell 2,3401 (2,737) UBS Securities Australia Ltd Sell 2,459 (476) UBS Securities Australia Ltd Sell 4,259 (476) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,656 1,656 UBS Securities Australia Ltd Buy 1,657 1,89 UBS Securities Australia Ltd Buy 1,677 1,89	-Mar-14	I IRS Securities Australia 11d	200	177,00	4 P	Gulnary
UBS Securities Australia Ltd Buy 142,183 (5,000) UBS Securities Australia Ltd Buy 36,489 10,000 UBS Securities Australia Ltd Buy 26,689 10,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 2,053 2,412 UBS Securities Australia Ltd Sell 23,001 (2,737) UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Buy 1,656 1,656 UBS Securities Australia Ltd Buy 1,657 188	Mar 14	1 IDC Cocurition A metallia 1+4	900	000,24	000,0	Ordinary
UBS Securities Australia Ltd Buy 142,183 15,622 UBS Securities Australia Ltd Sell 142,183 16,622 UBS Securities Australia Ltd Buy 86,689 10,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 20,632 2,412 UBS Securities Australia Ltd Sell 20,322 (2,437) UBS Securities Australia Ltd Sell 4,259 (475) UBS Securities Australia Ltd Sell 4,259 (476) UBS Securities Australia Ltd Sell 4,259 (476) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,656 158 UBS Securities Australia Ltd Buy 1,677 188	1	I IDC Commission A contract	IIAC C	45,030	(000,0)	Ordinary
UBS Securities Australia Ltd Sell 36,484 (4,554) UBS Securities Australia Ltd Buy 56,089 10,000 UBS Securities Australia Ltd Buy 2,5680 3,000 UBS Securities Australia Ltd Buy 2,0636 2,412 UBS Securities Australia Ltd Sell 2,03,32 (2,437) UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 2,259 (495) UBS Securities Australia Ltd Sell 3,415 (399) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,677 1,656 UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,677 1,88	- INIGI - I T	OBS Securities Australia Ltd	Buy	142,183	15,622	Ordinary
UBS Securities Australia Ltd Buy 86,089 10,000 UBS Securities Australia Ltd Sell 81,671 (9,000 UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Sell 2,012 UBS Securities Australia Ltd Sell 2,03322 (24,437) UBS Securities Australia Ltd Sell 2,259 (457) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,656 1,656 UBS Securities Australia Ltd Buy 1,657 188 UBS Securities Australia Ltd Buy 1,677 188	-Mar-14	UBS Securities Australia Ltd	Sell	364,484	(42,554)	Ordinary
UBS Securities Australia Ltd Sell 81 671 (9,500) UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 20,636 2,412 UBS Securities Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 4,259 (4,737) UBS Securities Australia Ltd Sell 4,759 (476) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,650 1,93 UBS Securities Australia Ltd Buy 1,607 1,88 UBS Securities Australia Ltd Buy 1,677 188	-Mar-14	UBS Securities Australia Ltd	Buy	86,089	10,000	Ordinary
UBS Securities Australia Ltd Buy 25,680 3,000 UBS Securities Australia Ltd Buy 20,636 2,412 UBS Securities Australia Ltd Sell 2,93,22 (2,437) UBS Wealth Management Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 4,259 (497) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,656 1,656 UBS Securities Australia Ltd Buy 1,657 188 UBS Securities Australia Ltd Buy 1,677 188	-Mar-14	UBS Securities Australia Ltd	Sell	81,671	(9,500)	Ordinary
UBS Securities Australia Ltd Buy 2,658 2,412 UBS Securities Australia Ltd Sell 2,09322 (24,437) UBS Securities Australia Ltd Sell 2,09322 (24,737) UBS Securities Australia Ltd Sell 4,259 (497) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,4159 1,656 UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,650 188	-Mar-14	UBS Securities Australia Ltd	Buy	25,680	3,000	Ordinary
UBS Securities Australia Ltd Sell 209,322 (2,437) UBS William Management Australia Ltd Sell 4,259 (497) UBS Securities Australia Ltd Sell 4,259 (497) UBS Securities Australia Ltd Sell 3,415 (476) UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,677 188	-Mar-14	UBS Securities Australia Ltd	Buy	20,636	2,412	Ordinary
UBS Weath Management Australia Ltd Sell 23,401 (2,737) UBS Securities Australia Ltd Sell 4,259 (497) UBS Securities Australia Ltd Sell 3,415 (398) UBS Securities Australia Ltd Buy 1,656 1,656 UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,650 188	-Mar-14	UBS Securities Australia Ltd	Sell	209,322	(24,437)	Ordinary
UBS Securities Australia Ltd Sell 4,259 (497) UBS Securities Australia Ltd Sell 3,415 (399) UBS Securities Australia Ltd Buy 14,159 (476) UBS Securities Australia Ltd Buy 1,650 198 UBS Securities Australia Ltd Buy 1,607 188	-Mar-14	UBS Wealth Management Australia Ltd	Sell	23,401	(2,737)	Ordinary
UBS Securities Australia Ltd Sell 3,415 (399) UBS Securities Australia Ltd Buy 1,4159 1,656 UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,650 193	-Mar-14	UBS Securities Australia Ltd	Sell	4.259	(497)	Ordinary
UBS Securities Australia Ltd Sell 4,075 (476) UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,650 193 UBS Securities Australia Ltd Buy 1,677 188	-Mar-14	UBS Securities Australia Ltd	les	3 415	(360)	Ordinany
UBS Securities Australia Ltd Buy 14,159 1,566 UBS Securities Australia Ltd Buy 1,650 133 UBS Securities Australia Ltd Buy 1,607 188	-Mar-14	UBS Securities Australia Itd	Sel	4 075	(377)	Croining
UBS Securities Australia Ltd Buy 1,607 188	Mar-14	LIBS Securities Australia Ltd	, c. a	14 150	1 656	Ordinary
UBS Securities Australia Ltd Buy 1,607 188	Mar.14	LIBC Countition Australia 144	Son S	14,139	050,1	Ordinary
UBS Securities Australia Ltd Buy 1,607 1,807		COS Securities Australia Liu	puy	000'	25	220
	Mar-14					

10 May 14	The state of the s			747	/uedion/
18-Mar-14	LIBS Securities Australia Ltd	Buy	5 164	21.0	Quin in
-Mar-14	1	Reav	3 164	0/5	Ordinary
18-Mar-14	۳	Bir	2,5	2/0	Ordinary Ordinary
Mar. 14	+	Berry	012,0	/2/	Ordinary
18-Mar-14	T	600	476.2	200	Ordinary
18-Mar-14	LIRS Securities Australia 14d	Paris.	3,104	310	Orginary
18-Mar-14	I IRS Securities Australia 14d	Sec. 9	61/7	310	Ordinary
Mar-14	LIPS Son mittee Australia 144	ond one	5,233	210	Ordinary
18.Mar-14	I IRS Securities Australia Ltd	bay.	26,552	3,310	Ordinary
S-Mar-14	IBO Countries Australia Ltd	, and	to 1.00	370	Ordinary
10-IVIGI-14	Obsider Australia Ltd	buy	3,753	439	Ordinary
- IPIN-0	CBS Securities Australia Ltd	Buy	/8/	92	Ordinary
P-INIGI-14	UBS Securities Australia Ltd	Buy	37,466	4,382	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	445	55	Ordinary
S-Mar-14	UBS Securities Australia Ltd	Buy	7,370	862	Ordinary
B-Mar-14	UBS Securities Australia Ltd	Buy	12,816	1,499	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	445	52	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	436	51	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	5,433	(634)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	209	71	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	4,822	564	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	5,069	242	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	3,523	412	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,505	176	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,479	173	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	5,309	270	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	1,915	224	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	659'8	428	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,601	688	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	3,651	427	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	231	27	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	4,044	473	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	878'E	450	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,223	143	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	727	85	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	868	105	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	3,198	374	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	4,053	474	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	3,540	414	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,693	198	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	5,045	290	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,368	160	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	6,737	788	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	4,437	519	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	4,788	260	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	4.489	525	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	4,446	520	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	812	95	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,360	276	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	4,908	574	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	4,506	527	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,274	149	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,445	169	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	941	110	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,975	348	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,855	217	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	11,842	1,385	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	10,978	1,284	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	686	115	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	6,823	798	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,077	126	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,975	348	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	3,454	404	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Birv	210		
l			oñ	37	Ordinary
18-Mar-14	II IRS Securities Australia Ltd	Bis	3.454	37	Ordinary

8-Mar-14	INV SOCIETIES ALISTRALIA TO		0301	70.	
8-Mar-14	UBS Securities Australia I to	Bus	3 506	410	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	14,963	1.750	Ordinary
4	UBS Securities Australia Ltd	Buv	2,514	294	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	259 22	2 647	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Biv	16.561	1 937	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buv	1.214	142	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	8.550	1 000	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	3,164	370	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	4,600	538	Ordinary
14	UBS Securities Australia Ltd	Buy	3,155	369	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	14,740	1,724	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	16,390	1,917	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	21,999	2,573	Ordinary
14	UBS Securities Australia Ltd	Buy	1,094	128	Ordinary
14	UBS Securities Australia Ltd	Buy	4,215	493	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	727	85	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	868	105	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	188	103	Ordinary
14	UBS Securities Australia Ltd	Buy	693	81	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,796	327	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	4,554	(532)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	5,805	679	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	744	87	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	274	32	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	21,119	2.470	Ordinary
-14	UBS Securities Australia Ltd	Buy	3,352	392	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	5,378	629	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	12,534	1,466	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	11,551	1,351	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	20,289	2,373	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,368	160	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	8,439	286	Ordinary
4	UBS Securities Australia Ltd	Buy	3,164	370	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Виу	16,903	1,977	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	12,398	1,450	Ordinary
4	UBS Securities Australia Ltd	Buy	7,755	206	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	8,434	282	Ordinary
4	UBS Securities Australia Ltd	Buy	17,158	2,008	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	197	23	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	787	95	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	11,645	1,362	Ordinary
4	UBS Securities Australia Ltd	Buy	6	-	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	22,256	2,603	Ordinary
4	UBS Securities Australia Ltd	Buy	3,164	370	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	3,164	370	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	119,511	1,627	Ordinary
4	UBS Securities Australia Ltd	Buy	4,275	200	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	13,176	1,541	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	56,285	6,583	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	20,614	2,411	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	299	20	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	862'25	6,760	Ordinary
4	UBS Securities Australia Ltd	Buy	2,779	325	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	5/6	114	Ordinary
4	UBS Securities Australia Ltd	Buy	1,616	189	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	282	33	Ordinary
8-Mar-14	Securities	Buy	2,967	347	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	856	112	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	826	112	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	826	112	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	828	112	Ordinary
	UBS Securities Australia Ltd	Buy	4,010	469	Ordinary
	UBS Securities Australia Ltd	Buy	7.114	750	
				760	Ordinary
18-Mar-14	ILIBS Securities Australia Ltd	Suv	2 043	232	Ordinary

6-ivlar-14	1100 Commission A	(5)	0.20/2		
18-Mar-14	I IRS Securities Australia I I d	Duy.	2,026	237	Ordinary
18-Mar-14	II IRS Societies Australia 14d	Bully	2,020	757	Ordinary
18-Mar-14	I IBS Societies Australia 14d	Bush	2,010	180	Ordinary
18 Mar 14	Life Counties Australia Ltd	buy.	010'	68	Ordinary
10-Mar 14	Upo comittee Australia Ltd	, and	660	0/	Ordinary
- IN	Obsideration of the Control of the C	anà	7,214	657	Ordinary
o-ividr-14	UBS Securities Australia Ltd	Buy	919'1	189	Ordinary
10-Wdr-14	Opp Securities Australia Ltd	Buy	248	59	Ordinary
10 Mar 14	Obsideration Australia Ltd	an)	600,1	81	Ordinary
10-IVIGIT-14	Obsideration Australia Ltd	griy	624	/3	Ordinary
16-Mar-14	UBS Securities Australia Ltd	guy	376	44	Ordinary
18-Mar-14	Sed T	Buy	205	24	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	770	90	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	291	34	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,847	216	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,907	223	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,907	223	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,907	223	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,915	224	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,009	235	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,898	222	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,890	221	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,813	212	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,248	146	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	928	112	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,069	242	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,958	525	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	727	85	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	795	66	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	7,148	836	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	6.395	748	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	3 005	(351)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	2 422	(283)	Ording.
18-Mar-14	LIBS Sequities Australia Ltd	Riv	103	13	Ordinary.
18-Mar-14	LIRS Securities Australia I to	Riv	501	203	Ordinary
8-Mar-14	I IRS Societies Australia 144	Coll	0,000	1010	Ciciliary
t Linds	OBS Seconties Australia Liu	nac d	1/0'9	(01.)	Ordinary
10-Mar-14	UBS Securities Australia Ltd	Sell	781	(91)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	117,769	(13,726)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	31,797	3,705	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	44,667	5,206	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	513	(09)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	17	(2)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Seil	566	(116)	Ordinary
8-Mar-14	UBS Securities Australia I td	3	1 975	(231)	idenibao
8-Mar-14	UBS Securities Australia I to	روا	0.52.5	(317)	Ordinary
18-Mar-14	LIBC Countries Australia and	11.10	000,0	(0.47)	Gulai
0-Mai-14	UBS Securities Australia Ltd	ank	451,//1	22,///	Ordinary
10-IVIGIT-14	ODS SECULTIES AUSTRAINA LICE	ânà	3//	44	Ordinary
0-1vial-14	Upo Committee Australia Liu	ond o	3,047	476	Orginary
o-Mar-14	UBS Securities Australia Lito	Buy	514	09	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,568	300	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	4,280	200	Ordinary
8-Mar-14	IUBS Securities Australia Ltd	Buy	4,280	200	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,568	300	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	3,082	360	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	21,494	2,511	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	22,222	2,596	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	4,280	200	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	471	55	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buv	1019	119	Ordinary
8-Mar-14	IUBS Securities Australia Ltd	Buv	2217	259	Ordinan
8-Mar-14	UBS Securities Australia (1d	Briv	928	110	Ordinary
18-Mar-14	I IRS Contrition Ametralia 14d	No.	0/6	1 6	Oldliary
8-Mar-14	I IDC Cocurities Australia 1+d	Dia.	030 1	200	Oldinary
0 140, 14	Tipo con trivial Australia Lita	boo	Oco,/	/16	Ordinary
0-IVId! - 4	ODS SECULIES MUSICALIA	hod	190,1	124	Ordinary
8-Mar-14	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		11111		
1	IOBS Securities Australia Ltd	anà	2,739	320	Ordinary

18-Mar-14	U65 Securities Australia Ltd	Buy	368	160	Ordinary
	1 DC Commission A section 1 and			100	
19 Mar 14	UBS Coomition America 14d	buy.	4,014	6.	Crainary
401 14	Upo comition Academic Ltd	Boy	1,5/7	10	Ordinary
10-IVIGI - 14	Obs Securities Australia Ltd	Buy	£9/'9	16/	Ordinary
7 - 10	UBS Securities Australia Ltd	Buy	1,428	/91	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	4,241	496	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,368	160	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,736	203	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	5,882	889	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,309	270	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buv	1.830	214	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	3,309	387	Ordinary
8-Mar-14	UBS Securities Australia I to	BIN	3 360	УОС	Ordinan
18-Mar-14	I IBS Securities Australia 14d	62	1 100	900	Ordinary
Apr. 1A	I IDC Consister Australia Ltd	600	:01,0	670	O'GIIGIN
- Indian	Opp Secontiles Australia Ltd	had	161.00	80.5	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	1,736	203	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	992	116	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	4,566	534	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,291	798	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	8,191	958	Ordinary
Mar-14	UBS Securities Australia Ltd	Buv	2.693	315	Ordinary
Mar-14	UBS Securities Australia 14d	Six	2,693	315	Ordinan
8-Mar-14	I IRS Societies Australia I to	600	3535		Ordinary
NA TEN O	1 Ibe Convision Assetsolic ted	, and	2,0/2	500	Claritary
1 1 1	Opp Securities Australia Ltd	anà	/09'1	88	Ordinary
0-IVIAI-14	Obs securities Australia Ltd	gay	//0′1	971	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,574	301	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,855	217	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	2,668	312	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	829	97	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buv	3.292	385	Ordinary
8-Mar-14	UBS Securities Australia 11d	B. v	239	82	Ordinary
8-Mar-14	I IBC Congrition Australia (4d	6.00	003 0	27	O'dilaily
Mar 14	Libe Countries Americal tel	100	555,2	/67	O'G'III'AII'A
1	COD Deculties Australia Ltd	ani	0,410	100	Orginary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,565	300	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,249	263	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,266	265	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,214	529	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buv	2.214	259	Ordinary
/ar-14	UBS Securities Australia Ltd	Buy	2 2 1 4	250	Ordinary
Aar.14	I IRS Societies Australia 14d	, ,	1100	030	Coding
	CO COUNTY AND	ind.	412,2	607	Granary
Adr-14	ODS Securities Australia Ltd	Buy	2,214	697	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,240	262	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,214	522	Ordinary
Aar-14	UBS Securities Australia Ltd	Buy	2,214	259	Ordinary
Aar-14	UBS Securities Australia Ltd	Bliv	2 24N	292	Ordinary
Apr.14	PRC Securities Australia Ltd	650	776 0	202	Countries
ti like	CDS Securities Australia Liu	śng	7,727	8/7	Ordinary
7ar-14	UBS Securities Australia Ltd	Buy	2,214	529	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,197	257	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,197	257	Ordinary
/ar-14	UBS Securities Australia Ltd	Buy	2.189	526	Ordinary
8-Mar-14	LIBS Securities Australia Ltd	Bix	2 120	8//0	Ordinan
Apr. 14	I IBS Societies Australia 14d	600	2 130	250	Ordinary
10 Mar 14	1 10C Commission American Ltd.	buy.	2,138	25.5	Crairialy
1 .	Opp Securities Australia Ltd	éna	167'7	807	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,138	250	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,283	267	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,214	528	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buv	2.026	237	Ordinary
8-Mar-14	UBS Securities Australia I td	Bir	2.018	236	Ordinan
	De Company of the Com	500	B10,2	222	Clairiary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,026	237	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,984	232	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	1,719	201	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buv	1,710	200	Ordinary
8-Mar-14	UBS Securities Australia I to	Bin	201	12	Ordinan
8-Mar-14	I IBC Securities Australia I to	, , ,	0000	20	Oudings
101-14	UBS Securities Australia Ltd	Buy	308	32	Ordinary
18-Mar-14	It IDC Consistion Assertalia 14d				,
	IOBS Securities Australia Liu	307	1,967	230	Ordinary

8-Var-14	UBS Securities Australia Ita	9	3.034	(354)	Ordinary
S-Mar 14	I IDO Commission Australia 142	11-3		11.1	
18-Mar 14	TIBS Countries Australia Ltd	Jell Cell	6 000	(E)	Ordinary
8-Mar-14	LIBS Counties Australia 144	100	2,025	(300)	Orginal
S Mar 14	TIDE Committee Australia Ltd	iac.	2,405	(288)	Ordinary
P Mar 14	UBS Securities Australia Ltd	196	7,037	(238)	Ordinary
O Mar 14	Libe committee Australia Lib) del	1/1	(DZ)	Ordinary
18.Mar.14	Libe contriber Australia ted	130	2,0,2	(235)	Ordinary
D Mar 14	Tibe contribe Australia Ltd	nac nac	1,943	(777)	Ordinar
18.Mar.17	LIBC Countries Australia Ltd	nac o	1 067	(68)	Crainary
8-Mar-14	Ť	361	/00/1	(171)	Ordinar
8-Mar-14	LIBS Countries Australia I to	361	1,542	(18U)	Ordinary
18-Mar-14	LIBS Societies Australia 144	100	0,710	(420)	Odgina
18-Mar-14	18C Societies Australia 14d	ll Soll	7 t, 10	(4/9)	Colonar
8-Mar-14	LIRS Societies Australia 144	ii o	2 107	(2704)	Orgina
No.	Library Administration Annual Control of the Contro	100	191,0	(3/3)	Ordinary
O-IVIGIT- 14	UBS Securities Australia Ltd	Sell	2,494	(291)	Ordinary
10-IVId! - 14	UBS Securities Australia Ltd	Sell	1,166	(136)	Ordinary
10-INIBI-14	UBS Securities Australia Ltd	Sell	2,756	(322)	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Sell	3,347	(391)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	625	(23)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	325	(38)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	2,174	(254)	Ordinar
8-Mar-14	UBS Securities Australia Ltd	Sell	3,938	(460)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	1,187	(138)	Ordina
18-Mar-14		Sell	396	(46)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	2,546	(562)	Ordinary
8-Mar-14	Г	Sell	2,675	(311)	Ordinary
8-Mar-14	Г	Sell	3,246	(377)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	4.051	(470)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	les	1,250	(145)	Ordinar
18-Mar-14	UBS Securities Australia Ltd	Sell	3.189	(370)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	1,112	(129)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	2.862	(332)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Stock returned	N/A	(250.000)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Buv	5.403	630	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Buy	87.424	10.225	Ording
9-Mar-14	I IBS Securities Australia 114	le o	202.200	(32,01)	500
O-Mar-14	۲	iac i	10101	(52,01)	Ordinary
10 Mar 14	100 Counties Australia Ltd	haa	1,050,475	122,235	Ordinary
O May 14	+	Jac.	052,500,1	(117,411)	Ordinary
9-May 14	Ups Securities Australia Ltd	Jian S	33,268	(3,8/3)	Ordinary
A-Mar-14	UBS Securities Australia Ltd	Jes	43,040	(2,000)	Ordinary
9-Mar- 14		Buy	256,043	29,686	Ordinary
9-Mar-14	UBS Global Asset Management	Buy	2,467,984	286,142	Ordinary
9-Mar-14	UBS Global Asset Management	Buy	2,110,761	244,725	Ordinary
9-Mar-14	UBS Global Asset Management	Buy	698,264	80,974	Ordinary
9-Mar-14	UBS Global Asset Management	Buy	1,874,553	217,382	Ordinary
9-Mar-14	UBS Global Asset Management	Buy	1,706,666	197,913	Ordinary
19-Mar-14	UBS Global Asset Management	Buy	399,905	46,609	Ordinary
19-Mar-14	UBS Global Asset Management	Buy	139,501	16,153	Ordinary
9-Mar-14	UBS Global Asset Management	Buy	150,511	17,454	Ordinary
9-Mar-14	UBS Global Asset Management	Buy	499,057	57,873	Ordinary
19-Mar-14	UBS Wealth Management Australia Ltd	Buy	2,769	322	Ordinary
9-Mar-14	UBS Wealth Management Australia Ltd	Buy	9,580	1,114	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Seil	3,504	(407)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Sell	421	(49)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Sell	1,428	(166)	Ordinan
19-Mar-14	UBS Securities Australia Ltd	Sell	3,062	(356)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Sell	3,672	(427)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Sell	4,968	(575)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Sell	4,838	(290)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Seli	130	(15)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Sell	090'19	(650'2)	Ordinary
9-Mar-14	UBS Securities Australia Ltd	Sell	112,146	(13,010)	Ordinar
9-Mar-14	UBS Securities Australia Ltd	Sell	4,491	(521)	Ordinary
10.Mar 14				(17)	5
	It IDC Commission Assembly 14d		14.253	(1 000)	

18-Mar-14	UBS Securities Australia Ltd	Buy	4,939	577	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	813	95	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	98	10	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	531	29	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	31,467	3,6	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	5,478		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	21.965	2	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	437	51	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	11.513		Ordinary
18-Mar-14	HRS Sequities Australia 11d	Biny	7.67.3		in in in
18-Mar-14	188 Securities Australia 14d	Bury	775		Ordinary
Name of	For Contract Australia 140	D.O.	417		Oldinary
10-Mdr-14	USS SECUTIVES AUSTRAIN LIQ	ani	69,345	8,101	Ordinary
10-IVIAI - 14	Ubs securities Australia Ltd	any	300		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,712	7	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	163	19	Ordinary
	UBS Securities Australia Ltd	Buy	382		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buv	3.159		Ordinary
	UBS Securities Australia Ltd	Buy	977 €		Ordinary
1	I IRS Securities Australia 14d	Biry	5110	001	Ordinary
	I DC Contribute Australia 14d	600	000		Oldlinary
to Ivial - It	ממס הבתו וופא אתאוושוש דות	ána.	650		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	3,159		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	368		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	111	13	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	2,303	592	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buv	52.156	9	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buv	368		Ordinary
18-Mar-14	UBS Securities Australia Hd	S. S.	137.7		Ordinan
18-Mar-14	I IBS Securities Australia 11d	Bin	E15 141		Ordinary
10 14 10	Day of the Assessment of the Control	boo.	14,141	00,100	Ordinary
10-IVId) - 14	DDS Securities Australia Ltd	griy	09/1/2		Ordinary
Io-Mdr-↓4	Ubs securities Australia Ltd	Buy	1,764		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	3,159		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1,909	223	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1.712		Ordinary
18-Mar-14	UBS Securities Australia Ltd	Buy	1 558	182	Ordinary
1	LIRS Securities Australia I to	N. G.	000 8	1 042	Carling
. _	1 IDC Consistion Australia 14d	a de	0,920	240'1	Ciginaly
П	The Country Action and	oci,	650		Ordinary
ı	Obs securities Australia Liu	ána	67/171	14,229	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	5,213		Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	13,568	1,5	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	623		Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	85 600		Ordinary
8-Mar-14	PR Securities Australia 11d	B. n.	9776		Carding
10 1425 14	1 DC Convision Australia 14.1	ora,	64/16	420	Clumary
0-IV:4I-14	Ubs securities Australia Ltd	Buy	3,159	369	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	779		Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	3,159		Ordinary
8-Mar-14	UBS Securities Australia Ltd	8uv	909'19	7,197	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Btiv	09		Ordinary
8-Mar-14	bit electricity Application 14d	Bind	2 150	050	Ordinary.
- IBIA-0	COO Decar lines Australia Etc	puy	901,0	505	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	890	104	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	21,717	2,537	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Buy	2,337	273	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Biv	86	C.	Ordinary
19 Mar 14	POC Commission A contraction 14d	, a	200 1	2 000	0.00
- Inida	ODS Securities Australia Liu	ána	7,625	330	Grainary
8-Mar-14	UBS Securities Australia Ltd	Sell	4,925	(576)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	6,592	(177)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	5,130	(009)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	1,104	(129)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	4.856	(268)	Ordinary
18-Mar-14	UBS Securities Australia I to	l e	478	(UEV)	Ordinan
8-Mar-14	IBS Societies Australia 14d	II o	F 173	(202)	Ordinary
0 1404 14	UDS Countries Australia (+d	201	0,170	(C0G)	Ordinary
lo-ivial 14	ODD Securities Australia Ltd	Je l	31/	(37)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Sell	3,261	(381)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	240	(82)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Sell	874	(102)	Ordinary
8-Mar-14	I IRS Sequirities Australia 11d	119	009 0	(167)	Ordinan
	CO CONTRACTOR AND		ESO.'F	() C+)	Ordinary
8-Mar-14	UBS Securities Australia Ltd	Zeii	4,/9/	(195)	Ordinary
18-Mar-14	UBS Securities Australia Ltd	Se!	4,172	(488)	Ordinary
Į					

UBS Securities Australia Ltd	Buy	18,767	2,195	Ordinary
UBS Securities Australia Ltd	Buv	11,513	1,336	Ordinary
UBS Securities Australia Ltd	Buy	868	105	Ordinary
JBS Securities Australia Ltd	Buy	35,448	4,146	Ordinary
UBS Securities Australia Ltd	Buy	6,327	740	Ordinary
JBS Securities Australia Ltd	, J	575	(07)	Ordinary
UBS Securities Australia Ltd	Sell	137	(16)	Ordinary
ities Australia Ltd	Sell	3,334	(389)	Ordinary
Securities Australia Ltd	Sell	1,106	(129)	Ordinary
UBS Securities Australia Ltd	Sell	746	(82)	Ordinary
BS Securities Australia Ltd	Jac.	1//	(06)	Ordinary
ities Australia Ltd	B	1,094	(177)	Ordinary
UBS Securities Australia Ltd	Sell	633	(77)	Ordinary
ities Australia Ltd	Sell	1,635	(191)	Ordinary
UBS Securities Australia Ltd	Setl	2,371	(277)	Ordinary
ities Australia Ltd	Sell	4,340	(202)	Ordinary
JBS Securities Australia Ltd	Buy	4,540	531	Ordinary
UBS Securities Australia Ltd	Buy	10,534	1,232	Ordinary
rities Australia Ltd	Sell	3,867	(451)	Ordinary
USS Securities Australia Ltd	Sell	0 1	(L)	Ordinary
rities Australia Ltd	Sell Sell	16/12	(175)	Ordinary
UBS Securities Australia Ltd	3	1 175	(737)	Ordinary
Australia Branch	Sell	33.625	(3 913)	Ordinary
UBS AG London Branch	Iss	65,952	(7,667)	Ordinary
JBS AG London Branch	Buy	65,275	7,667	Ordinary
JBS Securities Australia Ltd	Stock borrowed	NA	6,362	Ordinary
UBS Securities Australia Ltd	Stock returned	N/A	(82,000)	Ordinary
JBS Securities Australia Ltd	Stock returned	N/A	(38,631)	Ordinary
DBS Securities Australia Ltd	Stock returned	AN.	(80,000)	Ordinary
IRS Securities Australia Ltd	Stock returned	Z Z	(158,579)	Ordinary
JBS Securities Australia Ltd	Stock returned	AN AN	(2,000)	Ordinary
urities Australia Ltd	ē	AW	(115,915)	Ordinary
JBS Securities Australia Ltd	Stock returned	NA	(6,460)	Ordinary
JBS Securities Australia Ltd	Stock returned	N/A	(2,000)	Ordinary
UBS Securities Australia Ltd	Stock returned	NA	(30,000)	Ordinary
rrities Australia Ltd	Stock returned	NA	(6,349)	Ordinary
Securities Australia Ltd	Stock returned	NA	(82,786)	Ordinary
JBS Securities Australia Ltd	Stock returned	NA	(33,758)	Ordinary
JBS Securities Australia Ltd	Stock returned	NA	(000'09)	Ordinary
JBS Securities Australia Ltd	Stock returned	W.A	(346,868)	Ordinary
Unities Australia Ltd		ΔN.	(5,000)	Ordinary
unities Australia Ltd	Stock returned	AN S	(6,194)	Ordinary
LIBS Securities Australia 11d	Stock returned	47 V	(5,000)	Ordinary
UBS Securities Australia 11d	Stock returned	AW	(19 550)	Ordinary
JBS Securities Australia I td	Stock returned	A/N	(000,2)	Ordinary
UBS Securities Australia Ltd	Stock returned	W/N	(6.324)	Ordinary
UBS Securities Australia Ltd	Stock returned	N/A	(5.844)	Ordinary
UBS Securities Australia Ltd		A/N	(20,000)	Ordinary
rities Australia Ltd	Buy	37,083	4,307	Ordinary
rities Australia Ltd	Buy	52,765	6,100	Ordinary
UBS Securities Australia Ltd	Buy	948,273	110,569	Ordinary
UBS Securities Australia Ltd	Sell	821,588	(95,774)	Ordinary
Securities Australia Ltd	Sell	65,952	(7,667)	Ordinary
UBS Securities Australia Ltd	iii Xeii	95,975	(11,159)	Ordinary
IRC Global Assat Management	you a	194,040	22,595	Ordinary
al Asset Management	Rin	520,035	50,781	Ordingery
UBS Global Asset Management	Buv	520.893	60 781	Ordinary
UBS Global Asset Management		222/222		5
	2	520,893	60.781	Ordinary

		5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	77 765 765 1,030	(9)	Ordinary Ordinary
			7,7	(89)	Ordinary
		S S S S S S S S S S S S S S S S S S S	1,030	(120)	Ordinan
		Sel Sel			CICILIAN
		SSE SSE	2.145	(250)	Ordinary
		Sell	929	(73)	Ordinary
		Sell	3,582	(417)	Ordinary
		Sell	1,534	(179)	Ordinary
		Sell	51	(9)	Ordinary
		Sell	154	(18)	Ordinary
		Sell	3,743	(437)	Ordinary
		_	94	(11)	Ordinary
		Sell	3,462	(404)	Ordinary
		Sell	1,388	(162)	Ordinary
		Sell	479	(99)	Ordinary
		Sell	3,090	(361)	Ordinary
		Sell	950	(111)	Ordinary
		Sell	17	(2)	Ordinary
		Sell	3,178	(370)	Ordinary
		Sell	258	(65)	Ordinary
		Sell	1,237	(144)	Ordinary
		Buy	288,232	33,500	Ordinary
		Buy	320,282	37,225	Ordinary
		Buy	58,654	608'9	Ordinary
		Suy	141,798	16,461	Ordinary
		Stock returned	NA	(20,000)	Ordinary
		Buy	24,858	2,889	Ordinary
		Buy	908,460	105,614	Ordinary
777 		Sell	974,953	(113,256)	Ordinary
		Sell	71,945	(8,361)	Ordinary
	pt.	Buy	162,670	18,903	Ordinary
\neg	3	Sell	8,615	(1,000)	Ordinary
\top		Enter Into Derivative (NA	150	Options
$\neg \neg$		Buy	101,659	11,735	Ordinary
╛		Buy	174,061	20,151	Ordinary
		Buy	2,022	235	Ordinary
-		Buy	972	113	Ordinary
21-Mar-14 UBS Wealth Management Australia Ltd		Buy	5,945	169	Ordinary
		Buy	19,331	2,247	Ordinary
21-Mar-14 UBS Wealth Management Australia Ltd		Buy	2,400	279	Ordinary
Ī		Buy	3,321	386	Ordinary
21-Mar-14 UBS Wealth Management Australia Ltd		Buy	25,379	2,950	Ordinary
Г	t Australia Ltd	Buy	10,565	1,228	Ordinary
1-Mar-14 UBS Wealth Managemer		Buy	11,502	1,337	Ordinary
Ē		Buv	7.468	868	Ordinary
Ť		Buv	13,559	1 576	Ordinary
F		Buy	1,695	197	Ordinary
Ī		Buv	4.302	202	Ordinary
f		Buy	12,604	1.465	Ordinary
NBS		Buv	3.063	356	Ordinary
Ť		Buv	16.862	1 960	Ordinary
Ī		Buv	2,581	300	Ordinary
21-Mar-14 UBS Wealth Management Australia Ltd		Buv	2.245	261	Ordinary
21-Mar-14 UBS Wealth Managemen		AN AN	3.751	436	Ordinary
-Mar-14 UBS Wealth Management Australia Ltd		Buy	27 149	3 155	Ordinary
Т		Buv	5.093	597	Ordinary
Т		Buv	9636	1 120	Ordinary
T		Buv	2,590	301	Ordinary
Ť		Buy	8 199	953	Ordinary
T		Riv	3 502	407	Ordinary
T	Ī	Buy	200,0	0000	Ordings
1-Mar-14 I IDC Month Management Acattaila tid	Ī	ouy.	000'0	300	Ordinary
T		Buy.	3,880	401	Ordinary
T	T	Buy	4,224	169	Ordinary
Ť	Ī	, and	502,62	7607	Ordinary
Z1-Mar-14 UBS Wealth Management Australia Ltd		Buy	18,764	2,181	Ordinary

G.	100	5,100		(115)
UBS Wealth Management Australia Ltd	lles	1,391	(162)	Ordinary
UBS Wealth Management Australia Ltd	Buy	4,458	519	Ordinary
urities Australia Ltd	Sell	103	(12)	Ordinary
UBS Securities Australia Ltd	- N	138	(16)	Ordinary
UBS Securities Australia Ltd	Sell	353	(41)	Ordinary
UBS Securities Australia Ltd	, sel	1,987	(231)	Ordinary
Urities Australia Ltd	Sei	834	(97)	Ordinary
UBS Securities Australia Ltd	Sei	482	(26)	Ordinary
UBS Securities Australia Ltd	Sell	43	(2)	Ordinary
UBS Securities Australia Ltd	Sell	4,608	(534)	Ordinary
UBS Securities Australia Ltd	3	4,988	(578)	Ordinary
UBS Securities Australia Ltd	Sell	1,339	(155)	Ordinary
urities Australia Ltd	Sell	2,264	(262)	Ordinary
UBS Securities Australia Ltd	Şe‡	902	(02)	Ordinary
Securities Australia Ltd	Self	2,044	(236)	Ordinary
JBS Securities Australia Ltd	Sell	2.174	(251)	Ordinary
IBS Sequeities Australia Ltd	Zoll Coll	10.518	(1 216)	Ording
IBC Countries Australia 14d	21.0	1001	11,410,	O'GILIAI Y
Californ Australia Ltd	ing.	001	7	Ordinary
curries Australia Ltd	Buy	132,261	15,433	Ordinary
JBS Securities Australia Ltd	Buy	6,376	744	Ordinary
JBS Securities Australia Ltd	Sell	20.662	(2.411)	Ordinary
Curities Australia I td	218	19 205	2 241	Ordinan
IDO Commission Automobile Leal		CON'6	172,2	Oldinaly
colunes Australia Cito	, and	44,957	2,247	Ordinary
UBS Securities Australia Ltd	Buy	92,676	6,730	Ordinary
curities Australia Ltd	Sell	437	(21)	Ordinary
curities Australia Ltd	Sell	746	(87)	Ordinary
UBS Securities Australia Ltd	Sell	1 397	(163)	Ordinary
Securities Australia 11d	lay.	2 558	(311)	Ordingo
IBC Countries Australia 14d		CCC 7	(2.5)	O Claring y
Control October 19	100	2,222	(170)	Ordinary
Obs securities Australia Ltd	Sell	10,001	(1,16/)	Ordinary
UBS Securities Australia Ltd	Sell	685	(80)	Ordinary
curities Australia Ltd	Sell	3,030	(354)	Ordinary
UBS Securities Australia Ltd		3,719	(434)	Ordinary
UBS Securities Australia Ltd	Sell	5,056	(200)	Ordinary
curities Australia Ltd	Sell	1,054	(123)	Ordinary
UBS Securities Australia Ltd	#ag	3,593	(419)	Ordinary
curities Australia Ltd	Sell	797	(56)	Ordinary
IBC Contrities Australia 14d	Coll	303 6	(100)	
Control A county and	195	050,5	(+0+)	Couriery
DOS SECUTIONS AUSTRAINA LICA	Sell	85/,1	(502)	Ordinary
UBS Securities Australia Ltd	Sell	3,443	(401)	Ordinary
curities Australia Ltd	Sell	2,986	(348)	Ordinary
UBS Securities Australia Ltd	Sett	2,996	(349)	Ordinary
Securities Australia Ltd	Sell	1 296	(151)	Ordinary
Contribut Assertation 140	1130	555,1	(10)	Olding)
ODS Securities Australia Lid	Jac.	833	(16)	Ordinary
curities Australia Ltd	Sell	489	(57)	Ordinary
JBS Securities Australia Ltd	Sell	472	(22)	Ordinary
UBS Securities Australia Ltd	Sell	2,231	(260)	Ordinary
UBS Securities Australia 11d	les	1 656	(103)	Ordinary
Ourities Australia ! 1d	Coll	1 847	010	Ordinary
	1.00	101	(C Z)	Cluliaty
curilles Australia Ltd) Sell	7,294	(797)	Ordinary
UBS Securities Australia Ltd	Seil	1,065	(124)	Ordinary
UBS Securities Australia Ltd	Seli	318	(37)	Ordinary
UBS Securities Australia Ltd	Sell	1,014	(118)	Ordinary
Securities Australia Ltd	Sell	1 255	(146)	Ordinary
urities Australia I td	9	1100	(128)	Ordinan
IDC Countries Australia 144	1120	1,100	(100)	Ordinary
Unities Australia Lio		1,701	(198)	Ordinary
Securities Australia Ltd	Sell	1,556	(181)	Ordinary
JBS Securities Australia Ltd	Sell	1,031	(120)	Ordinary
Securities Australia Ltd	Sell	1,117	(130)	Ordinary
JBS Securities Australia Ltd	Sell	2.886	(388)	Ordinary
bit citation Australia 14d	100	020	(113)	
milities Australia Liu	Jell Control	0/6	(611)	Crainary
Securities Australia Ltd	Sell	541	(63)	Ordinary
JBS Securities Australia Ltd	Sell	2,686	(313)	Ordinary
Societies Australia Ltd	11-1			,
Unities Australia Ltd			,	
	120	961	(77)	Ordinary

	UBS Wealth Management Australia Ltd	Day		7 7 7	
	OC MACANTA A CONTRACTOR A CONTR	RIN	10,745	000	Ordinary
	22 Wealth Management Australia Lio	Buy	5 282	614	Ordinary
	IRS Wealth Management Australia Ltd	B. w	11001	100.	O'Glindry O'Glindry
	23 Weard Management Australa Ltd	puy	170,11	1,481	Ordinary
	UBS Wealth Management Australia Ltd	Buy	2,504	291	Ordinary
	UBS Wealth Management Australia Ltd	Buy	28,976	3,368	Ordinary
	UBS Wealth Management Australia Ltd	Buy	4,362	202	Ordinary
	35 Wealth Management Australia Ltd	Buy	19,288	2,242	Ordinary
	35 Wealth Management Australia Ltd	Buy	9.704	1.128	Ordinary
	JBS Wealth Management Australia Ltd	Buv	6.659	774	Ordinary
1111	IRS Wealth Management Australia Hd	, , ,	1 053	100	
	OC Months Administration According to the	(ng	000,1	/77	Ordinary
11	DD3 Wealth Management Australia Ltd	Buy	15,064	1,/51	Ordinary
	UBS Wealth Management Australia Ltd	Buy	5,506	640	Ordinary
	35 Wealth Management Australia Ltd	Buy	15,813	1,838	Ordinary
21-Mar-14 UB	JBS Wealth Management Australia Ltd	Buy	5.394	627	Ordinary
f	JBS Wealth Management Australia 11d	B _{1.2}	6 2 2 2 2	7.7.V	, de dipo
T	IRS Wealth Management Attentage 14	000	6000	177	or in any
Т	20 World Management Authority to	Puri.	/90/E	1,120	Ordinary
21-ividi-14	Obs wealth management Australia Ltd	gnà	4,370	208	Ordinary
T	ss wedin Management Australia Ltd	Buy	1,111	904	Ordinary
┪	35 Wealth Management Australia Ltd	Buy	2,908	338	Ordinary
	JBS Wealth Management Australia Ltd	Buy	4,758	553	Ordinary
21-Mar-14 UB	JBS Wealth Management Australia Ltd	Buv	10,616	1 234	Ordinary
┪	IRS Wealth Management Australia 144		0,000	1,704	o de la constante de la consta
Ť	22 Weard Management Australia Ltd	Buy	9,541	601,1	Ordinary
1	JBS Wealth Management Australia Ltd	Buy	19,022	2,211	Ordinary
_	35 Wealth Management Australia Ltd	Buy	6,349	738	Ordinary
1-Mar-14 UB	JBS Wealth Management Australia Ltd	Buv	8.879	1.032	Ordinary
21-Mar-14 UB	JBS Wealth Management Australia Ltd	Riv	11 736	1 306	Ordinan
Ť	IRC Wealth Management Australia 14d	70	0.11	200,	1000
Ť	22 Wealth Mai layer liert. Australia Ltd.	And	3,788	438	Ordinary
T	UBS Wealth Management Australia Ltd	Buy	11,141	1,295	Ordinary
	S Wealth Management Australia Ltd	Buy	5,351	622	Ordinary
21-Mar-14 UB	UBS Wealth Management Australia Ltd	Buy	4216	70D7	Ordinary
Г	S Wealth Management Australia Ltd	Ben	787 6	100	Crainage
Т	Street, Management Australia and	Boy	2,404	504	Ordinary
1	UBS Wealth Management Australia Ltd	Buy	1,609	187	Ordinary
1-Mar-14 UB	UBS Wealth Management Australia Ltd	Buy	4,173	485	Ordinary
1-Mar-14 UB	UBS Wealth Management Australia Ltd	Buv	5.730	999	Ordinary
-Mar-14 UBS	S Wealth Management Australia Ltd	200	2 331	27.1	Ordinan
Ť	10C Month Management Australia 144	6	100,3	1,7	o diligit
Ť	o wealth wallagernern Australia Ltd	buy	6,418	/46	Orginary
1	UBS Wealth Management Australia Ltd	Виу	6,384	742	Ordinary
1-Mar-14 UB	UBS Wealth Management Australia Ltd	Buy	3,200	372	Ordinary
1-Mar-14 UB	UBS Wealth Management Australia Ltd	Sev.	8.302	965	Ordinary
-Mar-14 I.B	JBS Wealth Management Australia 11d	. B	033 0	300	
T	Manager A composition of the control	íng.	6000	066	Cidinaly
Ť	S wealth Management Australia Ltd	Buy	2,581	300	Ordinary
٦	JBS Wealth Management Australia Ltd	Buy	8,947	1,040	Ordinary
21-Mar-14 UB	UBS Wealth Management Australia Ltd	Buy	13,843	1,609	Ordinary
21-Mar-14 (UB)	UBS Wealth Management Australia Ltd	Buv	6.444	749	Ordinary
Ī	IBS Weekh Management Australia 14d	Paris,	2,52,5	24.7	(initial)
Ť	S vegici ivia iagerrent Australia Ltd	, ind	996,6	647	Ordinary
1	UBS Wealth Management Australia Ltd	Buy	6,280	730	Ordinary
	S Wealth Management Australia Ltd	Buy	3,054	355	Ordinary
1-Mar-14 UB:	UBS Wealth Management Australia Ltd	Buy	4,078	474	Ordinary
1-Mar-14 UB:	UBS Wealth Management Australia Ltd	Buv	2.650	308	Ordinary
1-Mar-14 I.B	IBS Wealth Management Australia Ltd	21.0	421 078	00000	Ordina
T	Chicago Managament Assessment	ína c	421,978	43,043	Ciginary
1	DDS Wealth Management Australia Ltd	Ang	13,6/9	1,590	Ordinary
1-Mar-14 UB:	S Wealth Management Australia Ltd	Buy	8,044	935	Ordinary
	UBS Wealth Management Australia Ltd	Buy	2,985	347	Ordinary
21-Mar-14 UB:	JBS Wealth Management Australia Ltd	Buy	2,925	340	Ordinary
21-Mar-14 UB:	UBS Wealth Management Australia Ltd	Buv	5.953	269	Ordinary
1-Mar-14 UB	UBS Wealth Management Australia Ltd	Riv	26.730	3 107	Ordinary
T	C Months & Annual Company Annual and		05/07	20,00	O C
1-iviar-14 OB.	DES Wealth Management Australia Ltd	gnà	CC5, \C	4,342	Ordinary
1	S Wealth Management Australia Ltd	Buy	2,779	323	Ordinary
21-Mar-14 UB	JBS Wealth Management Australia Ltd	Buy	3,252	378	Ordinary
21-Mar-14 UBS	UBS Wealth Management Australia Ltd	Buv	5.781	672	Ordinary
Ť	UBS Sequrities Australia 11d	9	20 530	(111)	Ordina
T	IRS Securities Australia 14d		062.61	, , , ,	1000
Ŧ	DOS CONTROL A CAST BIRD ELLO	ang.	65/,21	1/4/1	Ordinary
7	S Securities Austrana Ltd	Nell Nell	198,989	(8/6,72)	Ordinary
21-Mar-14 UBS	UBS Securities Australia Ltd	Buy	342,416	39,540	Ordinary
Ť	IDC Consisting Australia 14.4	(50)	014,240	0000	, in in

UBS Wealth Management Australia Ltd UBS Wealth Management Australi	ar-14	U65 Wealth Mahagement Australia Ltd	200	3.773	73/	Ordinary
USS Vealth Management Australia Ltd Buy 1,527	ar-14	I IRC Wasth Management Australia Ltd	Bir	E7 90E	6 710	, in the second
1925 Verbith Management Australia Ltd 1949 1,225 1,2		I IRS Wealth Management Australia Ind	Buy	17 671	2,054	Ordinary
Bis Wealth Management Australia Lid Buy 6,022 1,172 Bis Wealth Management Australia Lid Buy 4,4241 497 Bis Wealth Management Australia Lid Buy 4,5241 4,97 Bis Wealth Management Australia Lid Buy 4,6566 5,38 Bis Wealth Management Australia Lid Buy 4,6567 5,38 Bis Wealth Management Australia Lid Buy 4,6567 5,38 Bis Wealth Management Australia Lid Buy 4,6567 5,38 Bis Wealth Management Australia Lid Buy 6,53 5,40 Bis Wealth Management Australia Lid Buy 6,52 5,20 Bis Wealth Management Australia Lid Buy 7,52 6,62 Bis Wealth Management Australia Lid Buy 6,52 5,20 Bis Wealth Management Australia Lid Buy 6,52	ar-14	I IRC Wealth Management Australia 14d	200	1/0'/1	4004	Ordinary
USS Wealth Management Australia Ltd Buy 5,242 4,59 USS Wealth Management Australia Ltd Buy 4,355 5,69 USS Wealth Management Australia Ltd Buy 4,355 5,99 USS Wealth Management Australia Ltd Buy 4,565 5,38 USS Wealth Management Australia Ltd Buy 4,565 5,38 USS Wealth Management Australia Ltd Buy 4,565 5,38 USS Wealth Management Australia Ltd Buy 5,22 5,28 USS Wealth Management Australia Ltd Buy 4,565 5,23 6,9 USS Wealth Management Australia Ltd Buy 5,22 6,9 5,24 6,9 USS Wealth Management Australia Ltd Buy 5,24 6,0 5,24 6,9 USS Wealth Management Australia Ltd Buy 5,24 6,0 5,2 6,0 USS Wealth Management Australia Ltd Buy 5,24 6,0 6,0 5,2 6,0 6,0 6,0 6,0 6,0 6,0 6,0 6,0 6,0 6,0	17	LIBS Wealth Manageritetti Australia Ltd	buy.	9,003	227,1	Ordinary
USS Wealth Management Australia Ltd Buy 4,241 459 455	1	I IBC Woolff Management Australia Ind	, no	9,022	362	Ordinary
USS Wealth Management Australia Lid Buy 4,555 576 USS Wealth Management Australia Lid Buy 4,555 5,59 USS Wealth Management Australia Lid Buy 5,04 5,29 USS Wealth Management Australia Lid Buy 5,05 5,13 USS Wealth Management Australia Lid Buy 5,22 5,13 USS Wealth Management Australia Lid Buy 5,23 5,13 USS Wealth Management Australia Lid Buy 5,23 6,13 USS Wealth Management Australia Lid Buy 5,25 6,14 USS Wealth Management Australia Lid Buy 5,25 6,07 USS Wealth Management Australia Lid Buy 5,25 6,07 USS Wealth Management Australia Lid Buy 5,22 6,07 USS Wealth Management Australia Lid Buy 5,22 6,07 USS Wealth Management Australia Lid Buy 6,07 7,24 USS Wealth Management Australia Lid Buy 6,07 7,24 USS Wealth Management Australia Lid Buy 6,07 7,24 <td>17.75</td> <td>I BC Woolth Management Australia 14d</td> <td>900</td> <td>0,410</td> <td>760</td> <td>Ordinary</td>	17.75	I BC Woolth Management Australia 14d	900	0,410	760	Ordinary
USS Wealth Management Australia Ltd Buy 5,429 5,689 UBS Wealth Management Australia Ltd Buy 5,648 5,688 UBS Wealth Management Australia Ltd Buy 4,611 5,388 UBS Wealth Management Australia Ltd Buy 4,611 5,388 UBS Wealth Management Australia Ltd Buy 5,315 7,34 UBS Wealth Management Australia Ltd Buy 5,013 5,23 UBS Wealth Management Australia Ltd Buy 5,013 7,21 UBS Wealth Management Australia Ltd Buy 5,013 7,178 UBS Wealth Management Australia Ltd Buy 5,580 1,178 UBS Wealth Management Australia Ltd Buy 5,580 1,425 UBS Wealth Management Australia Ltd Buy 5,580 1,426 UBS Wealth Management Australia Ltd Buy 5,580 1,427 UBS Wealth Management Australia Ltd Buy 5,580 1,427 UBS Wealth Management Australia Ltd Buy 5,580 1,427 UBS Wealth Management Australia Ltd Buy 7,522<	ar-14	I IRS Wealth Management Australia I th	Birk	7 055	323	Ordinary
USS Wealth Management Australia Ltd Buy 45,686 5,899 USS Wealth Management Australia Ltd Buy 45,611 5,308 USS Wealth Management Australia Ltd Buy 4,611 5,315 1,145 USS Wealth Management Australia Ltd Buy 6,317 7,21 619 USS Wealth Management Australia Ltd Buy 6,317 7,21 610 USS Wealth Management Australia Ltd Buy 6,314 7,22 6,42 USS Wealth Management Australia Ltd Buy 1,42 6,42 1,72 USS Wealth Management Australia Ltd Buy 6,80 7,22 6,42 USS Wealth Management Australia Ltd Buy 6,80 7,22 6,43 1,72 USS Wealth Management Australia Ltd Buy 6,80 7,22 6,43 1,52 USS Wealth Management Australia Ltd Buy 6,80 7,52 6,43 1,52 USS Wealth Management Australia Ltd Buy 6,52 7,52 6,43 1,52 1,52 USS Wealth Management Australia Ltd <td>ar-14</td> <td>ILBS Wealth Management Australia Ltd</td> <td>200</td> <td>1 970</td> <td>000</td> <td>Ordinary</td>	ar-14	ILBS Wealth Management Australia Ltd	200	1 970	000	Ordinary
USS Wealth Namegement Australia Ltd Buy 4,556 5,308 USS Wealth Namegement Australia Ltd Buy 4,516 5,308 USS Wealth Namegement Australia Ltd Buy 9,811 1,445 USS Wealth Namegement Australia Ltd Buy 9,811 7,34 USS Wealth Namegement Australia Ltd Buy 5,335 1,72 USS Wealth Namegement Australia Ltd Buy 5,336 1,72 USS Wealth Namegement Australia Ltd Buy 2,013 2,01 USS Wealth Namegement Australia Ltd Buy 1,62 1,72 USS Wealth Namegement Australia Ltd Buy 2,52 64 USS Wealth Namegement Australia Ltd Buy 2,52 64 USS Wealth Namegement Australia Ltd Buy 2,52 64 USS Wealth Namegement Australia Ltd Buy 4,28 1,405 USS Wealth Namegement Australia Ltd Buy 2,52 1,407 USS Wealth Namegement Australia Ltd Buy 2,52 1,407 USS Wealth Namegement Australia Ltd Buy 2,52 1,	ar-14	I IRC Wealth Management Australia Ltd	Buy	50.402	070 3	Ordinary
USS Wealth Namagement Australia Ltd Bluy 4,511 USS Wealth Management Australia Ltd Bluy 4,511 USS Wealth Management Australia Ltd Bluy 6,315 USS Wealth Management Australia Ltd Bluy 6,315 USS Wealth Management Australia Ltd Bluy 7,227 USS Wealth Management Australia Ltd Bluy 7,227 USS Wealth Management Australia Ltd Bluy 7,227 USS Wealth Management Australia Ltd Bluy 7,227 6,42 USS Wealth Management Australia Ltd Bluy 7,227 6,43 USS Wealth Management Australia Ltd Bluy 7,227 6,43 USS Wealth Management Australia Ltd Bluy 7,227 6,43 USS Wealth Management Australia Ltd Bluy 6,524 7,275 USS Wealth Management Australia Ltd Bluy 7,275 7,275 USS Wealth Management Australia	114	I IDC Months Management Activities and	200	10,45¢	500,0	Colliary
USS Wealth Management Australia Ltd Buy 9,851 1,346 USS Wealth Management Australia Ltd Buy 9,851 1,346 USS Wealth Management Australia Ltd Buy 5,335 734 USS Wealth Management Australia Ltd Buy 5,335 1,726 USS Wealth Management Australia Ltd Buy 1,276 1,276 USS Wealth Management Australia Ltd Buy 1,427 1,226 USS Wealth Management Australia Ltd Buy 1,427 1,427 USS Wealth Management Australia Ltd Buy 9,145 1,427 USS Wealth Management Australia Ltd Buy 4,241 4,221 USS Wealth Management Australia Ltd Buy 4,241 4,241 USS Wealth Management Australia Ltd Buy 4,241 4,241 USS Wealth Management Australia Ltd Buy 4,241 4,241 USS Wealth Management Australia Ltd Buy 5,540 6,60 USS Wealth Management Australia Ltd Buy 7,241 4,231 USS Wealth Management Australia Ltd Buy 5,241<	1 2	I DC Woolsh Management Australia Ltd	âna	45,000	2,308	Orginary
UBS Wealth Management Australia Ltd Buy 6,381 1,135 UBS Wealth Management Astralia Ltd Buy 6,381 7,134 UBS Wealth Management Astralia Ltd Buy 5,225 619 UBS Wealth Management Astralia Ltd Buy 2,017 234 UBS Wealth Management Astralia Ltd Buy 1,047 1,178 UBS Wealth Management Astralia Ltd Buy 1,470 1,470 UBS Wealth Management Astralia Ltd Buy 1,470 4,821 UBS Wealth Management Astralia Ltd Buy 5,540 4,821 UBS Wealth Management Astralia Ltd Buy 5,540 4,821 UBS Wealth Management Astralia Ltd Buy 5,540 4,871 UBS Wealth Management Astralia Ltd Buy 5,540 4,821 UBS Wealth Management Astralia Ltd Buy 5,540 4,821 UBS Wealth Management Astralia Ltd Buy 5,540 4,821 UBS Wealth Management Astralia Ltd Buy 5,540 4,821 4,05 UBS Wealth Management Astralia Ltd Buy <th< td=""><td></td><td>ops wealth management Australia Liu</td><td>, and</td><td>110,4</td><td>950</td><td>Ordinary</td></th<>		ops wealth management Australia Liu	, and	110,4	950	Ordinary
UBS Wealth Management Astralia Ltd Buy 5,375 734 UBS Wealth Management Astralia Ltd Buy 5,973 734 UBS Wealth Management Astralia Ltd Buy 7,013 234 UBS Wealth Management Astralia Ltd Buy 7,014 7,015 UBS Wealth Management Astralia Ltd Buy 1,026 1,126 UBS Wealth Management Astralia Ltd Buy 9,653 1,122 UBS Wealth Management Astralia Ltd Buy 9,653 1,122 UBS Wealth Management Astralia Ltd Buy 5,556 644 UBS Wealth Management Astralia Ltd Buy 5,526 643 UBS Wealth Management Astralia Ltd Buy 5,526 643 UBS Wealth Management Astralia Ltd Buy 5,526 1,447 UBS Wealth Management Astralia Ltd Buy 5,527 8,20 UBS Wealth Management Astralia Ltd Buy 5,524 5,746 UBS Wealth Management Astralia Ltd Buy 5,249 7,70 UBS Wealth Management Astralia Ltd Buy 5,241 7,70	1	ODS Weditt Management Australia Ltd	áng	ICB/K	1,145	Ordinary
UBS Wealth Management Australia Ltd Buy 5,325 619 UBS Wealth Management Australia Ltd Buy 2,017 291 234 UBS Wealth Management Australia Ltd Buy 2,018 791 292 UBS Wealth Management Australia Ltd Buy 1,478 1,718 UBS Wealth Management Australia Ltd Buy 1,628 1,478 1,621 UBS Wealth Management Australia Ltd Buy 1,249 1,478 1,623 UBS Wealth Management Australia Ltd Buy 5,540 6,403 1,478 1,623 UBS Wealth Management Australia Ltd Buy 5,540 6,403 1,405 <	ar-14	UBS Wealth Management Australia Ltd	Buy	6,315	734	Ordinary
UBS Wealth Management Australia Ltd Buy 2,013 234 UBS Wealth Management Australia Ltd Buy 791 12,16 UBS Wealth Management Australia Ltd Buy 1,478 1,128 UBS Wealth Management Australia Ltd Buy 9,145 1,128 UBS Wealth Management Australia Ltd Buy 9,145 1,128 UBS Wealth Management Australia Ltd Buy 7,224 6,44 UBS Wealth Management Australia Ltd Buy 7,224 6,44 UBS Wealth Management Australia Ltd Buy 7,224 6,43 UBS Wealth Management Australia Ltd Buy 7,524 6,53 UBS Wealth Management Australia Ltd Buy 7,524 7,52 UBS Wealth Management Australia Ltd Buy 7,524	ar-14	UBS Wealth Management Australia Ltd	Buy	5,325	619	Ordinary
UBS Wealth Management Australia Ltd Buy 791 92 UBS Wealth Management Australia Ltd Buy 1,718 1,718 UBS Wealth Management Australia Ltd Buy 9,653 1,718 UBS Wealth Management Australia Ltd Buy 9,653 1,718 UBS Wealth Management Australia Ltd Buy 5,526 6,45 UBS Wealth Management Australia Ltd Buy 5,527 840 UBS Wealth Management Australia Ltd Buy 5,540 6,45 UBS Wealth Management Australia Ltd Buy 5,540 6,45 UBS Wealth Management Australia Ltd Buy 5,540 6,43 UBS Wealth Management Australia Ltd Buy 5,540 6,43 UBS Wealth Management Australia Ltd Buy 5,540 6,50 UBS Wealth Management Australia Ltd Buy 5,540 6,50 UBS Wealth Management Australia Ltd Buy 5,721 8,00 UBS Wealth Management Australia Ltd Buy 5,721 8,00 UBS Wealth Management Australia Ltd Buy 5,721	ar-14	UBS Wealth Management Australia Ltd	Buy	2,013	234	Ordinary
UBS Wealth Management Australia Ltd Bby 10,461 1,216 UBS Wealth Management Australia Ltd Bby 1,675 1,172 UBS Wealth Management Australia Ltd Bby 9,145 1,172 UBS Wealth Management Australia Ltd Bby 1,678 4,871 UBS Wealth Management Australia Ltd Bby 1,272 8.40 UBS Wealth Management Australia Ltd Bby 1,274 4,241 4,93 UBS Wealth Management Australia Ltd Bby 1,274 4,241 4,93 UBS Wealth Management Australia Ltd Bby 1,2,62 1,62 UBS Wealth Management Australia Ltd Bby 5,540 6,82 UBS Wealth Management Australia Ltd Bby 5,540 6,82 UBS Wealth Management Australia Ltd Bby 6,72 7,67 UBS Wealth Management Australia Ltd Bby 6,72 7,72 UBS Wealth Management Australia Ltd Bby 6,73 7,72 UBS Wealth Management Australia Ltd Bby 6,73 7,73 UBS Wealth Management Australia Ltd	ar-14	UBS Wealth Management Australia Ltd	Buy	791	92	Ordinary
UBS Wealth Management Australia Ltd Buy 14,780 1,718 UBS Wealth Management Australia Ltd Buy 91,475 1,063 UBS Wealth Management Australia Ltd Buy 91,475 1,063 UBS Wealth Management Australia Ltd Buy 91,475 4821 UBS Wealth Management Australia Ltd Buy 5,540 644 UBS Wealth Management Australia Ltd Buy 1,2,449 1,447 UBS Wealth Management Australia Ltd Buy 6,409 7,544 UBS Wealth Management Australia Ltd Buy 6,409 7,674 UBS Wealth Management Australia Ltd Buy 6,5409 7,674 UBS Wealth Management Australia Ltd Buy 6,609 7,741 UBS Wealth Management Australia Ltd Buy 6,609 7,741 UBS Wealth Management Australia Ltd Buy <td< td=""><td>ar-14</td><td>UBS Wealth Management Australia Ltd</td><td>Buy</td><td>10,461</td><td>1,216</td><td>Ordinary</td></td<>	ar-14	UBS Wealth Management Australia Ltd	Buy	10,461	1,216	Ordinary
UBS Wealth Management Australia Ltd Buy 9,633 1,122 UBS Wealth Management Australia Ltd Buy 9,145 1,063 UBS Wealth Management Australia Ltd Buy 7,227 8,40 UBS Wealth Management Australia Ltd Buy 7,227 8,40 UBS Wealth Management Australia Ltd Buy 7,227 8,40 UBS Wealth Management Australia Ltd Buy 7,244 9,147 UBS Wealth Management Australia Ltd Buy 7,244 9,241 UBS Wealth Management Australia Ltd Buy 6,805 7,51 UBS Wealth Management Australia Ltd Buy 6,243 7,00 UBS Wealth Management Australia Ltd Buy 6,243 7,00 UBS Wealth Management Australia Ltd Buy 6,242	21-Mar-14	UBS Wealth Management Australia Ltd	Buy	14.780	1.718	Ordinary
UBS Wealth Management Australia Ltd Buy 9,145 1,653 UBS Wealth Management Australia Ltd Buy 5,145 6,46 UBS Wealth Management Australia Ltd Buy 5,540 6,43 UBS Wealth Management Australia Ltd Buy 5,540 6,43 UBS Wealth Management Australia Ltd Buy 6,805 7,91 UBS Wealth Management Australia Ltd Buy 4,241 4,93 UBS Wealth Management Australia Ltd Buy 4,241 4,93 UBS Wealth Management Australia Ltd Buy 4,241 4,93 UBS Wealth Management Australia Ltd Buy 6,624 7,57 UBS Wealth Management Australia Ltd Buy 7,674 8,93 UBS Wealth Management Australia Ltd Buy 7,674 8,93 UBS Wealth Management Australia Ltd Buy 6,624 7,74 UBS Wealth Management Australia Ltd Buy 6,231 7,74 UBS Wealth Management Australia Ltd Buy 6,241 7,93 UBS Wealth Management Australia Ltd Buy 6,241	ar-14	UBS Wealth Management Australia Ltd	Buy	9,653	1,122	Ordinary
UBS Wealth Management Australia Ltd Buy 41,476 4,621 UBS Wealth Management Australia Ltd Buy 7,527 840 UBS Wealth Management Australia Ltd Buy 7,227 840 UBS Wealth Management Australia Ltd Buy 7,527 840 UBS Wealth Management Australia Ltd Buy 5,540 6,43 UBS Wealth Management Australia Ltd Buy 5,540 6,43 UBS Wealth Management Australia Ltd Buy 5,540 6,43 UBS Wealth Management Australia Ltd Buy 7,574 892 UBS Wealth Management Australia Ltd Buy 7,574 892 UBS Wealth Management Australia Ltd Buy 7,574 892 UBS Wealth Management Australia Ltd Buy 6,797 7,419 UBS Wealth Management Australia Ltd Buy 6,797 7,419 UBS Wealth Management Australia Ltd Buy 7,523 3,20 UBS Wealth Management Australia Ltd Buy 7,523 3,03 1,223 UBS Wealth Management Australia Ltd Buy	ar-14	UBS Wealth Management Australia Ltd	Buv	9.145	1.063	Ordinary
UBS Wealth Management Australia Ltd Buy 5,558 646 UBS Wealth Management Australia Ltd Buy 7,227 840 UBS Wealth Management Australia Ltd Buy 1,249 1,441 UBS Wealth Management Australia Ltd Buy 6,805 791 UBS Wealth Management Australia Ltd Buy 6,805 791 UBS Wealth Management Australia Ltd Buy 6,805 791 UBS Wealth Management Australia Ltd Buy 6,805 704 UBS Wealth Management Australia Ltd Buy 6,824 7,746 UBS Wealth Management Australia Ltd Buy 6,827 7,70 UBS Wealth Management Australia Ltd Buy 6,83 7,71 UBS Wealth Management Australia Ltd Buy 6,83 7,71 UBS Wealth Management Australia Ltd Buy 6,83 <t< td=""><td>ar-14</td><td>UBS Wealth Management Australia Ltd</td><td>Buy</td><td>41 476</td><td>4 821</td><td>Ordinary</td></t<>	ar-14	UBS Wealth Management Australia Ltd	Buy	41 476	4 821	Ordinary
UBS Wealth Management Australia Ltd Buy 7,227 840 UBS Wealth Management Australia Ltd Buy 12,449 1,447 UBS Wealth Management Australia Ltd Buy 12,449 1,447 UBS Wealth Management Australia Ltd Buy 6,895 731 UBS Wealth Management Australia Ltd Buy 6,895 731 UBS Wealth Management Australia Ltd Buy 6,895 731 UBS Wealth Management Australia Ltd Buy 6,895 734 UBS Wealth Management Australia Ltd Buy 6,797 754 892 UBS Wealth Management Australia Ltd Buy 6,797 736 1,605 1,405 UBS Wealth Management Australia Ltd Buy 6,797 1,700 1,208 1,405 1,508 1,509 <td< td=""><td>ar-14</td><td>I IBS Wealth Management Australia Ltd</td><td>Blay</td><td>2 22 2</td><td>30'5</td><td>Colorary</td></td<>	ar-14	I IBS Wealth Management Australia Ltd	Blay	2 22 2	30'5	Colorary
UBS Wealth Management Australia Ltd Buy 17,424 1,447 UBS Wealth Management Australia Ltd Buy 1,244 1,447 UBS Wealth Management Australia Ltd Buy 5,240 6,805 751 UBS Wealth Management Australia Ltd Buy 6,805 751 105 UBS Wealth Management Australia Ltd Buy 6,805 751 105 UBS Wealth Management Australia Ltd Buy 6,624 7,674 825 UBS Wealth Management Australia Ltd Buy 6,624 7,746 105 UBS Wealth Management Australia Ltd Buy 2,752 105 105 UBS Wealth Management Australia Ltd Buy 5,721 6,624 7,70 UBS Wealth Management Australia Ltd Buy 2,753 3,20 1,527 UBS Wealth Management Australia Ltd Buy 2,753 3,20 1,527 1,20 UBS Wealth Management Australia Ltd Buy 2,753 3,20 1,22 1,22 1,22 UBS Wealth Management Australia Ltd Buy 2,753 </td <td>1</td> <td>I IBC Wealth Management Australia</td> <td>900</td> <td>7 222</td> <td>2 6</td> <td>o diament</td>	1	I IBC Wealth Management Australia	900	7 222	2 6	o diament
UBS Wealth Management Australia Ltd Buy 1,2,443 1,128 UBS Wealth Management Australia Ltd Buy 5,540 644 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 6,805 791 UBS Wealth Management Australia Ltd Buy 6,602 735 UBS Wealth Management Australia Ltd Buy 1,502 1,502 UBS Wealth Management Australia Ltd Buy 1,502 1,502 UBS Wealth Management Australia Ltd Buy 6,624 7,70 UBS Wealth Management Australia Ltd Buy 6,624 7,70 UBS Wealth Management Australia Ltd Buy 6,729 7,71 UBS Wealth Management Australia Ltd Buy 6,729 7,72 UBS Wealth Management Australia Ltd Buy 1,523 1,229 UBS Wealth Management Australia Ltd Buy 6,642 7,77 UBS Wealth Management Australia Ltd Buy 6,642 7,77 UBS Wealth Management Australia Ltd Buy 6,642	11.74	I IBC Woolth Management Australia 140	600	127,	040	Countries
UBS Wealth Management Australia Ltd Buy 6.75 UBS Wealth Management Australia Ltd Buy 5.540 6.43 UBS Wealth Management Australia Ltd Buy 6.805 791 UBS Wealth Management Australia Ltd Buy 6.805 791 UBS Wealth Management Australia Ltd Buy 6.624 7.76 UBS Wealth Management Australia Ltd Buy 6.624 7.76 UBS Wealth Management Australia Ltd Buy 6.73 7.76 UBS Wealth Management Australia Ltd Buy 6.73 7.76 UBS Wealth Management Australia Ltd Buy 6.73 7.70 UBS Wealth Management Australia Ltd Buy 6.73 7.70 UBS Wealth Management Australia Ltd Buy 6.73 7.73 UBS Wealth Management Australia Ltd Buy 6.73 7.71		I IDC Woulth Management Australia 144	600	644,21	1,1	Gullary
UBS Wealth Management Australia Ltd Buy 5,240 643 UBS Wealth Management Australia Ltd Buy 4,241 4,34 UBS Wealth Management Australia Ltd Buy 6,805 731 UBS Wealth Management Australia Ltd Buy 7,674 802 UBS Wealth Management Australia Ltd Buy 7,362 1,522 UBS Wealth Management Australia Ltd Buy 7,362 1,405 UBS Wealth Management Australia Ltd Buy 2,726 1,522 UBS Wealth Management Australia Ltd Buy 2,735 7,70 UBS Wealth Management Australia Ltd Buy 1,208 1,229 UBS Wealth Management Australia Ltd Buy 1,227 2,24 UBS Wealth Management Australia Ltd Buy 1,227 2,24 UBS Wealth Management Australia Ltd Buy 1,227 2,24 UBS Wealth Management Australia Ltd Buy 6,387 7,419 UBS Wealth Management Australia Ltd Buy 6,387 7,419 UBS Wealth Management Australia Ltd Buy 6,387	1	Obs. Wealth Mailagenrent Australia Ltd.	eny	10,023	1,236	Orolnary
UBS Wealth Management Australia Ltd Buy 4,241 433 UBS Wealth Management Australia Ltd Buy 6,409 745 UBS Wealth Management Australia Ltd Buy 7,674 892 UBS Wealth Management Australia Ltd Buy 7,674 892 UBS Wealth Management Australia Ltd Buy 4,345 5,04 UBS Wealth Management Australia Ltd Buy 6,624 7,70 UBS Wealth Management Australia Ltd Buy 6,737 1,405 UBS Wealth Management Australia Ltd Buy 6,737 1,605 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 6,542 772 UBS Wealth Management Australia Ltd Buy 4,241 4,241 UBS Wealth Management Australia Ltd Buy 6,585	6	Obstance Indianagement Australia Liu	pnà	0,540	44	Ordinary
UBS Wealth Management Australia Ltd Buy 6,805 791 UBS Wealth Management Australia Ltd Buy 6,805 791 UBS Wealth Management Australia Ltd Buy 7,674 892 UBS Wealth Management Australia Ltd Buy 7,674 892 UBS Wealth Management Australia Ltd Buy 23,624 2,746 UBS Wealth Management Australia Ltd Buy 23,624 2,746 UBS Wealth Management Australia Ltd Buy 4,335 1,522 UBS Wealth Management Australia Ltd Buy 6,737 790 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 6,622 771 UBS Wealth Management Australia Ltd Buy 6,624 775 UBS Wealth Management Australia Ltd Buy 6,737 371 UBS Wealth Management Australia Ltd Buy 6,737 7419 UBS Wealth Management Australia Ltd Buy 6,737 <td< td=""><td>91-14</td><td>UBS Wealth Management Australia Ltd</td><td>Buy</td><td>4,241</td><td>493</td><td>Ordinary</td></td<>	91-14	UBS Wealth Management Australia Ltd	Buy	4,241	493	Ordinary
UBS Wealth Management Australia Ltd Buy 7,649 842 UBS Wealth Management Australia Ltd Buy 7,644 862 UBS Wealth Management Australia Ltd Buy 7,644 562 UBS Wealth Management Australia Ltd Buy 23,624 770 UBS Wealth Management Australia Ltd Buy 6,624 770 UBS Wealth Management Australia Ltd Buy 6,737 708 UBS Wealth Management Australia Ltd Buy 6,737 709 UBS Wealth Management Australia Ltd Buy 2,731 665 UBS Wealth Management Australia Ltd Buy 1,927 2,241 493 UBS Wealth Management Australia Ltd Buy 1,527 2,241 493 UBS Wealth Management Australia Ltd Buy 1,527 2,241 433 UBS Wealth Management Australia Ltd Buy 6,642 7,712 UBS Wealth Management Australia Ltd Buy 6,542 7,712 UBS Wealth Management Australia Ltd Buy 6,523 7,64 UBS Wealth Management Australi	ar-14	UBS Wealth Management Australia Ltd	Buy	6,805	791	Ordinary
UBS Wealth Management Australia Ltd Buy 7,674 B82 UBS Wealth Management Australia Ltd Buy 4,345 505 UBS Wealth Management Australia Ltd Buy 23,624 1,746 UBS Wealth Management Australia Ltd Buy 23,624 1,746 UBS Wealth Management Australia Ltd Buy 5,721 665 UBS Wealth Management Australia Ltd Buy 6,737 709 UBS Wealth Management Australia Ltd Buy 2,753 224 UBS Wealth Management Australia Ltd Buy 2,753 224 UBS Wealth Management Australia Ltd Buy 4,241 433 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 4,241 433 UBS Wealth Management Australia Ltd Buy 6,643 772 UBS Wealth Management Australia Ltd Buy 6,643 772 UBS Wealth Management Australia Ltd Buy 6,573 7419 UBS Wealth Management Australia Ltd Buy 4,281 5	ar-14	UBS Wealth Management Australia Ltd	Buy	6,409	745	Ordinary
UBS Wealth Management Australia Ltd Buy 4.345 505 UBS Wealth Management Australia Ltd Buy 13,094 1,522 UBS Wealth Management Australia Ltd Buy 6,624 770 UBS Wealth Management Australia Ltd Buy 4,335 1,405 UBS Wealth Management Australia Ltd Buy 4,328 5,156 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 3,261 387 UBS Wealth Management Australia Ltd Buy 4,241 433 UBS Wealth Management Australia Ltd Buy 5,822 7419 UBS Wealth Management Australia Ltd Buy 6,327 771 UBS Wealth Management Australia Ltd Buy 6,327 771 UBS Wealth Management Australia Ltd Buy 4,775 585 UBS Wealth Management Australia Ltd Buy 4,775 582 UBS Wealth Management Australia Ltd Buy 4,775	ar-14	UBS Wealth Management Australia Ltd	Buy	7,674	892	Ordinary
UBS Wealth Management Australia Ltd Buy 13,094 1,522 UBS Wealth Management Australia Ltd Buy 5,121 66.24 770 UBS Wealth Management Australia Ltd Buy 6,624 770 UBS Wealth Management Australia Ltd Buy 6,727 665 UBS Wealth Management Australia Ltd Buy 6,737 709 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 3,751 370 UBS Wealth Management Australia Ltd Buy 3,751 373 UBS Wealth Management Australia Ltd Buy 3,733 3,871 UBS Wealth Management Australia Ltd Buy 6,542 772 UBS Wealth Management Australia Ltd Buy 4,775 5859 UBS Wealth Management Australia Ltd Buy 4,775 5859 UBS Wealth Management Australia Ltd Buy 4,775 5859 UBS Wealth Management Australia Ltd Buy	lar-14	UBS Wealth Management Australia Ltd	Buy	4.345	505	Ordinary
UBS Wealth Management Australia Ltd Buy 23,624 2,746 UBS Wealth Management Australia Ltd Buy 6524 770 UBS Wealth Management Australia Ltd Buy 6,524 770 UBS Wealth Management Australia Ltd Buy 6,737 759 UBS Wealth Management Australia Ltd Buy 2,753 320 UBS Wealth Management Australia Ltd Buy 2,753 320 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 4,241 439 UBS Wealth Management Australia Ltd Buy 6,642 771 UBS Wealth Management Australia Ltd Buy 6,642 771 UBS Wealth Management Australia Ltd Buy 6,538 551 UBS Wealth Management Australia Ltd Buy 4,735 397 UBS Wealth Management Australia Ltd Buy 4,735 397 UBS Wealth Management Australia Ltd Buy 4,735 555 UBS Wealth Management Australia Ltd Buy 4,381 579 <td>ar-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Buy</td> <td>13.094</td> <td>1 522</td> <td>Ordinary</td>	ar-14	UBS Wealth Management Australia Ltd	Buy	13.094	1 522	Ordinary
UBS Wealth Management Australia Ltd Buy 6,624 770 UBS Wealth Management Australia Ltd Buy 6,624 770 UBS Wealth Management Australia Ltd Buy 5,721 665 UBS Wealth Management Australia Ltd Buy 2,733 320 UBS Wealth Management Australia Ltd Buy 2,733 320 UBS Wealth Management Australia Ltd Buy 3,261 433 UBS Wealth Management Australia Ltd Buy 3,221 433 UBS Wealth Management Australia Ltd Buy 3,233 3,333 UBS Wealth Management Australia Ltd Buy 6,642 7,712 UBS Wealth Management Australia Ltd Buy 6,523 7,719 UBS Wealth Management Australia Ltd Buy 6,523 7,719 UBS Wealth Management Australia Ltd Buy 6,523 7,719 UBS Wealth Management Australia Ltd Buy 6,233 7,64 UBS Wealth Management Australia Ltd Buy 6,233 7,719 UBS Wealth Management Australia Ltd Buy 6,233	ar-14	LIBS Wealth Management Australia Ltd	Rim	23 624	2746	Ordinan
UBS Wealth Management Australia Ltd Buy 1,000 UBS Wealth Management Australia Ltd Buy 5,721 665 UBS Wealth Management Australia Ltd Buy 6,737 700 UBS Wealth Management Australia Ltd Buy 6,737 730 UBS Wealth Management Australia Ltd Buy 6,737 730 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 6,642 7,19 UBS Wealth Management Australia Ltd Buy 6,642 7,19 UBS Wealth Management Australia Ltd Buy 6,573 7,419 UBS Wealth Management Australia Ltd Buy 5,859 661 UBS Wealth Management Australia Ltd Buy 4,281 5,79 UBS Wealth Management Australia Ltd Buy 4,281 5,91 UBS Wealth Management Australia Ltd Buy 4,281 5,91	25.14	I IBS Woolth Management Australia Ltd	Bush	120,62	07.7	Coming
UBS Wealth Management Australia Ltd Buy 4,335 1,405 UBS Wealth Management Australia Ltd Buy 4,335 5,156 UBS Wealth Management Australia Ltd Buy 2,753 320 UBS Wealth Management Australia Ltd Buy 2,753 320 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 6,53,27 7,419 UBS Wealth Management Australia Ltd Buy 6,53,27 7,419 UBS Wealth Management Australia Ltd Buy 6,53,27 7,419 UBS Wealth Management Australia Ltd Buy 4,775 5,85 UBS Wealth Management Australia Ltd Buy 4,775 5,82 UBS Wealth Management Australia Ltd Buy 4,78 5,51 UBS Wealth Management Australia Ltd Buy 6,73 7,77 UBS Wealth Management Australia Ltd Buy 6,78	1,1	Libo Months Management Author Co.	25.0	12,088	20,	Oldinary
UBS Wealth Management Australia Ltd Buy 4,215 Bbb UBS Wealth Management Australia Ltd Buy 6,737 790 UBS Wealth Management Australia Ltd Buy 2,753 320 UBS Wealth Management Australia Ltd Buy 4,241 463 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 6,542 772 UBS Wealth Management Australia Ltd Buy 6,542 772 UBS Wealth Management Australia Ltd Buy 4,775 585 UBS Wealth Management Australia Ltd Buy 4,775 585 UBS Wealth Management Australia Ltd Buy 4,775 585 UBS Wealth Management Australia Ltd Buy 6,289 754 UBS Wealth Management Australia Ltd Buy 6,289 754 UBS Wealth Management Australia Ltd Buy 6,289 754		100 West of the transport of the least of th	bod.	12,088	1,403	Oldinary
UBS Wealth Management Australia Ltd Buy 4.358 5,156 UBS Wealth Management Australia Ltd Buy 6,793 790 UBS Wealth Management Australia Ltd Buy 2,753 224 UBS Wealth Management Australia Ltd Buy 4,241 493 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 6,573 7419 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,573 764 </td <td>1 1</td> <td>Obs wealth Management Australia Ltd</td> <td>ing</td> <td>17/'5</td> <td>599</td> <td>Ordinary</td>	1 1	Obs wealth Management Australia Ltd	ing	17/'5	599	Ordinary
UBS Wealth Management Australia Ltd Buy 6,797 790 UBS Wealth Management Australia Ltd Buy 2,753 320 UBS Wealth Management Australia Ltd Buy 1,927 224 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 3,267 1,323 UBS Wealth Management Australia Ltd Buy 6,642 7,72 UBS Wealth Management Australia Ltd Buy 6,642 7,72 UBS Wealth Management Australia Ltd Buy 6,642 7,72 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,775 558 UBS Wealth Management Australia Ltd Buy 4,735 555 UBS Wealth Management Australia Ltd Buy 4,735 551 UBS Wealth Management Australia Ltd Buy 4,736 552 UBS Wealth Management Australia Ltd Buy 5,317 618 UBS Wealth Management Australia Ltd Buy 4,581 579	4	UBS Wealth Management Australia Ltd	Buy	44,358	5,156	Ordinary
UBS Wealth Management Australia Ltd Buy 1,753 320 UBS Wealth Management Australia Ltd Buy 1,927 224 UBS Wealth Management Australia Ltd Buy 4,241 439 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 6,573 751 UBS Wealth Management Australia Ltd Buy 6,573 750	ar-14	JUBS Wealth Management Australia Ltd	Buy	6,797	790	Ordinary
UBS Wealth Management Australia Ltd Buy 4,241 424 UBS Wealth Management Australia Ltd Buy 4,241 439 UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 6,622 777 UBS Wealth Management Australia Ltd Buy 6,63,827 7,419 UBS Wealth Management Australia Ltd Buy 6,829 681 UBS Wealth Management Australia Ltd Buy 4,775 582 UBS Wealth Management Australia Ltd Buy 4,775 582 UBS Wealth Management Australia Ltd Buy 4,775 552 UBS Wealth Management Australia Ltd Buy 4,785 552 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,733 764 UBS Wealth Management Australia Ltd Buy 6,733 6,89 UBS Wealth Management Australia Ltd Buy 6,289 6,79 UBS Wealth Management Australia Ltd Buy 6,781 7	ar-14	UBS Wealth Management Australia Ltd	Buy	2,753	320	Ordinary
UBS Wealth Management Australia Ltd Buy 4,241 493 UES Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 33,303 3,873 UBS Wealth Management Australia Ltd Buy 65,42 772 UBS Wealth Management Australia Ltd Buy 63,822 7,719 UBS Wealth Management Australia Ltd Buy 4,775 5859 UBS Wealth Management Australia Ltd Buy 4,775 585 UBS Wealth Management Australia Ltd Buy 4,775 585 UBS Wealth Management Australia Ltd Buy 4,775 582 UBS Wealth Management Australia Ltd Buy 6,283 754 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 6,381 779 UBS Wealth Management Australia Ltd Buy 6,381 7	I-Mar-14	UBS Wealth Management Australia Ltd	Buy	1,927	224	Ordinary
UBS Wealth Management Australia Ltd Buy 3,261 379 UBS Wealth Management Australia Ltd Buy 3,261 373 UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 5,859 681 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 3,415 397 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,573 79 UBS Wealth Management Australia Ltd Buy 6,289 73 UBS Wealth Management Australia Ltd Buy 6,317 6,89 UBS Wealth Management Australia Ltd Buy 6,317 6,99 UBS Wealth Management Australia Ltd Buy 6,317 9,31 <td>ar-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Buv</td> <td>4.241</td> <td>493</td> <td>Ordinary</td>	ar-14	UBS Wealth Management Australia Ltd	Buv	4.241	493	Ordinary
LIES Wealth Management Australia Ltd Buy 1,223 LIES Wealth Management Australia Ltd Buy 33,309 3,871 LIES Wealth Management Australia Ltd Buy 6,642 772 LIES Wealth Management Australia Ltd Buy 6,642 772 LIES Wealth Management Australia Ltd Buy 5,859 681 LIES Wealth Management Australia Ltd Buy 4,775 555 LIES Wealth Management Australia Ltd Buy 4,775 555 LIES Wealth Management Australia Ltd Buy 4,835 552 LIES Wealth Management Australia Ltd Buy 6,289 754 LIES Wealth Management Australia Ltd Buy 6,289 754 LIES Wealth Management Australia Ltd Buy 6,289 754 LIES Wealth Management Australia Ltd Buy 6,391 579 LIES Wealth Management Australia Ltd Buy 6,391 579 LIES Wealth Management Australia Ltd Buy 6,391 570 LIES Wealth Management Australia Ltd Buy 6,573 750	21.14	I RS Wealth Management Australia Ltd	Bury	2 261	02.6	Ordinary
UBS Wealth Management Australia Ltd Buy 1,382 3,871 UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 6,642 775 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 6,573 3415 397 UBS Wealth Management Australia Ltd Buy 6,573 764 108 108 1,288 108 108 108 108 108 1,288 108 108 108 108 1,288 108 1,288 108 1,288 108 1,288 108 1,288 108 1,288 108 108 1,288 108 1,288 108 1,288 108 1,288 108 1,288 1,288 1		Library and agentic to Australia and Library	, and	197'6	6/6	Ordinary
UBS Wealth Management Australia Ltd Buy 3.871 UBS wealth Management Australia Ltd Buy 6.642 772 UBS wealth Management Australia Ltd Buy 6.642 772 UBS Wealth Management Australia Ltd Buy 5.859 5.85 UBS Wealth Management Australia Ltd Buy 4,775 5.55 UBS Wealth Management Australia Ltd Buy 4,835 5.62 UBS Wealth Management Australia Ltd Buy 6,573 7.64 UBS Wealth Management Australia Ltd Buy 6,289 7.79 UBS Wealth Management Australia Ltd Buy 6,289 7.79 UBS Wealth Management Australia Ltd Buy 6,289 7.79 UBS Wealth Management Australia Ltd Buy 4,381 5.79 UBS Wealth Management Australia Ltd Buy 4,381 9.79 UBS Wealth Management Australia Ltd Buy 8,371 9.73 UBS Wealth Management Australia Ltd Buy 5,370 6.68 UBS Wealth Management Australia Ltd Buy 5,570 6.67 <td>ar-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Buy</td> <td>11,382</td> <td>1,323</td> <td>Ordinary</td>	ar-14	UBS Wealth Management Australia Ltd	Buy	11,382	1,323	Ordinary
UBS Wealth Management Australia Ltd Buy 6,642 772 UBS Wealth Management Australia Ltd Buy 63,827 7,419 UBS Wealth Management Australia Ltd Buy 5,859 681 UBS Wealth Management Australia Ltd Buy 5,859 681 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,317 618 UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 5,317 6570 UBS Wealth Management Australia Ltd Buy 5,317 6570 UBS Wealth Management Australia Ltd Buy 5,570 309	ar-14	UBS Wealth Management Australia Ltd	Вчу	33,303	3,871	Ordinary
UBS Wealth Management Australia Ltd Buy 63,827 7,419 UBS Wealth Management Australia Ltd Buy 1,229 1,229 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,573 79 UBS Wealth Management Australia Ltd Buy 6,573 79 UBS Wealth Management Australia Ltd Buy 5,317 6,18 UBS Wealth Management Australia Ltd Buy 8,317 9,31 UBS Wealth Management Australia Ltd Buy 8,31 9,31 UBS Wealth Management Australia Ltd Buy 8,31 9,31 UBS Wealth Management Australia Ltd Buy 2,500 9,01 UBS Wealth Management Australia Ltd Buy 2,500 30	ar-14	UBS Wealth Management Australia Ltd	Buy	6,642	772	Ordinary
UBS Wealth Management Australia Ltd Buy 10,573 1,229 UBS Wealth Management Australia Ltd Buy 4,775 5859 681 UBS Wealth Management Australia Ltd Buy 4,775 5555 682 UBS Wealth Management Australia Ltd Buy 4,775 5552 682 UBS Wealth Management Australia Ltd Buy 6,573 3415 397 UBS Wealth Management Australia Ltd Buy 6,289 731 618 UBS Wealth Management Australia Ltd Buy 4,981 5,317 618 UBS Wealth Management Australia Ltd Buy 5,317 618 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 2,500 670 UBS Wealth Management Australia Ltd Buy 2,500 670 UBS Wealth Management Australia Ltd Buy 1,9865 2,309 <td>ar-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Buy</td> <td>63.827</td> <td>7.419</td> <td>Ordinary</td>	ar-14	UBS Wealth Management Australia Ltd	Buy	63.827	7.419	Ordinary
UBS Wealth Management Australia Ltd Buy 5,855 681 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 6,573 754 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 4,981 5,79 UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 8,317 9,531 UBS Wealth Management Australia Ltd Buy 8,317 9,531 UBS Wealth Management Australia Ltd Buy 8,317 9,531 UBS Wealth Management Australia Ltd Buy 5,670 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,500 3	ar-14	I IRS Wealth Management Australia (14	Birv	10.573	1 220	Ordinary
UBS Wealth Management Australia Ltd Buy 3,053 UBS Wealth Management Australia Ltd Buy 4,775 558 UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 6,389 731 UBS Wealth Management Australia Ltd Buy 5,317 618 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 2,5670 659 UBS Wealth Management Australia Ltd Buy 2,5670 670 UBS Wealth Management Australia Ltd Buy 6,075 7,309 <t< td=""><td></td><td>I IDC Months Administration of Australia Lead</td><td>600</td><td>0.000</td><td>1,622.3</td><td>Oldinary Coldinary</td></t<>		I IDC Months Administration of Australia Lead	600	0.000	1,622.3	Oldinary Coldinary
USB Wealth Management Australia Ltd Buy 4,775 555 UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 6,573 754 UBS Wealth Management Australia Ltd Buy 6,289 779 UBS Wealth Management Australia Ltd Buy 6,289 779 UBS Wealth Management Australia Ltd Buy 6,317 618 UBS Wealth Management Australia Ltd Buy 8,317 973 UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 5,670 309 UBS Wealth Management Australia Ltd Buy 5,670 309 UBS Wealth Management Australia Ltd Buy 6,675 2,309 <td></td> <td>USC WESTER INSTRUMENT AUSTRALIS ETC.</td> <td>Buy</td> <td>600'0</td> <td>8 1</td> <td>Ordinary</td>		USC WESTER INSTRUMENT AUSTRALIS ETC.	Buy	600'0	8 1	Ordinary
UBS Wealth Management Australia Ltd Buy 1,288 UBS Wealth Management Australia Ltd Buy 4,835 552 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 5,317 6,18 UBS Wealth Management Australia Ltd Buy 8,317 9,32 UBS Wealth Management Australia Ltd Buy 8,317 9,31 UBS Wealth Management Australia Ltd Buy 8,371 9,31 UBS Wealth Management Australia Ltd Buy 5,670 5,670 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 1,9865 2,309 UBS Wealth Management Australia Ltd Buy 5,670 301 UBS Wealth Management Australia Ltd Buy 6,675 2,309 UBS Wealth Management Australia Ltd Buy 6,675 7,509 <td>ar-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Buy</td> <td>4,775</td> <td>222</td> <td>Ordinary</td>	ar-14	UBS Wealth Management Australia Ltd	Buy	4,775	222	Ordinary
UBS Wealth Management Australia Ltd Buy 4,835 562 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 5,317 618 UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 5,670 5,309 UBS Wealth Management Australia Ltd Buy 5,016 5,309 UBS Wealth Management Australia Ltd Buy 6,016 5,309 UBS Wealth Management Australia Ltd Buy 6,016 5,	ar-14	UBS Wealth Management Australia Ltd	Buy	11,081	1,288	Ordinary
UBS Wealth Management Australia Ltd Buy 3,415 397 UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 4,981 5/9 UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 8,317 9,73 UBS Wealth Management Australia Ltd Buy 8,371 9,73 UBS Wealth Management Australia Ltd Buy 5,670 509 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,500 301 UBS Wealth Management Australia Ltd Buy 4,557 1,090 UBS Wealth Management Australia Ltd Buy 5,016 5,016 UBS Wealth Management Australia Ltd Buy 4,557 1,090	ar-14	UBS Wealth Management Australia Ltd	Buv	4.835	295	Ordinary
UBS Wealth Management Australia Ltd Buy 6,573 764 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 4,981 5/9 UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 81,911 9,521 UBS Wealth Management Australia Ltd Buy 5,670 559 UBS Wealth Management Australia Ltd Buy 5,670 559 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 5,016 539 UBS Wealth Management Australia Ltd Buy 4,455 1,099	ar-14	UBS Wealth Management Australia Ltd	Buv	3.415	795	Ordinary
UBS Wealth Management Australia Ltd Buy 6,279 784 UBS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 4,981 579 UBS Wealth Management Australia Ltd Buy 3,390 334 UBS Wealth Management Australia Ltd Buy 8,171 9,521 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 6,016 5,317 UBS Wealth Management Australia Ltd Buy 5,016 5,319 UBS Wealth Management Australia Ltd Buy 6,016 5,016 UBS Wealth Management Australia Ltd Buy 6,016 5,016 UBS Wealth Management Australia Ltd Buy 6,016 <t< td=""><td>1 1 4</td><td>1 IDC Months Management Australia Ltd</td><td>P. C.</td><td>0.110</td><td>100</td><td>Countries</td></t<>	1 1 4	1 IDC Months Management Australia Ltd	P. C.	0.110	100	Countries
USS Wealth Management Australia Ltd Buy 6,289 731 UBS Wealth Management Australia Ltd Buy 4,981 579 UBS Wealth Management Australia Ltd Buy 5,317 618 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 2,5670 659 UBS Wealth Management Australia Ltd Buy 19,865 2,309 UBS Wealth Management Australia Ltd Buy 5,016 5,01 UBS Wealth Management Australia Ltd Buy 5,016 5,01 UBS Wealth Management Australia Ltd Buy 5,016 5,01 UBS Wealth Management Australia Ltd Buy 4,455 1,799 UBS Wealth Management Australia Ltd Buy 4,455 1,799	1	ODS Wedith Management Australia Ltd	ana	0,5/3	/64	Ordinary
UBS Wealth Management Australia Ltd Buy 4,981 579 UBS Wealth Management Australia Ltd Buy 5,317 618 UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 8,711 973 UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 5,016 5,309 UBS Wealth Management Australia Ltd Buy 5,016 5,309 UBS Wealth Management Australia Ltd Buy 5,016 5,016 UBS Wealth Management Australia Ltd Buy 5,016 5,016 UBS Wealth Management Australia Ltd Buy 4,455 1,099	91-14	UBS Wealth Management Australia Ltd	Buy	6,289	/31	Ordinary
UBS Wealth Management Australia Ltd Buy 5.317 618 UBS Wealth Management Australia Ltd Buy 3.390 394 UBS Wealth Management Australia Ltd Buy 8,371 9,73 UBS Wealth Management Australia Ltd Buy 8,371 9,73 UBS Wealth Management Australia Ltd Buy 2,500 301 UBS Wealth Management Australia Ltd Buy 1,965 2,309 UBS Wealth Management Australia Ltd Buy 5,016 301 UBS Wealth Management Australia Ltd Buy 5,016 5,016 UBS Wealth Management Australia Ltd Buy 4,655 1,099 UBS Wealth Management Australia Ltd Buy 4,655 1,099	ar-14	UBS Wealth Management Australia Ltd	Buy	4,981	579	Ordinary
UBS Wealth Management Australia Ltd Buy 3,390 394 UBS Wealth Management Australia Ltd Buy 8,1911 9,521 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 5,670 559 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 5,016 539 UBS Wealth Management Australia Ltd Buy 5,016 539 UBS Wealth Management Australia Ltd Buy 5,016 539 UBS Wealth Management Australia Ltd Buy 5,016 5,016	ar-14	UBS Wealth Management Australia Ltd	Buy	5,317	618	Ordinary
UBS Wealth Management Australia Ltd Buy 81,911 9,521 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 19,865 2,309 UBS Wealth Management Australia Ltd Buy 5,016 581 UBS Wealth Management Australia Ltd Buy 4,455 1,099	ar-14	UBS Wealth Management Australia Ltd	Burv	3 390	394	Ordinary
USE Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 8,371 973 UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 5,016 5,309 UBS Wealth Management Australia Ltd Buy 5,016 5,309 UBS Wealth Management Australia Ltd Buy 6,016 5,016 UBS Wealth Management Australia Ltd Buy 4,455 1,099	ar-14	II IRS Wealth Management Australia Ltd	Rin	81 911	0 521	Ordinary
USF Wealth Management Australia Ltd Buy 5,670 5973 UBS Wealth Management Australia Ltd Buy 5,670 509 UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 19,865 2,309 UBS Wealth Management Australia Ltd Buy 5,016 583 UBS Wealth Management Australia Ltd Buy 4,455 1,099		100 March Management Automobile 144	final	116,10	120,5	Gollary
UBS Wealth Management Australia Ltd Buy 5,670 659 UBS Wealth Management Australia Ltd Buy 2,500 301 UBS Wealth Management Australia Ltd Buy 19,865 2,309 UBS Wealth Management Australia Ltd Buy 5,016 583 UBS Wealth Management Australia Ltd Buy 4,55 1,090	+	Ubb wealth Management Australia Ltd	Buy	8,3/1	9/3	Ordinary
UBS Wealth Management Australia Ltd Buy 2,590 301 UBS Wealth Management Australia Ltd Buy 19,865 2,309 UBS Wealth Management Australia Ltd Buy 5,016 583 UBS Wealth Management Australia Ltd Buy 4,455 1,090	ar-14	UBS Wealth Management Australia Ltd	Buy	5,670	629	Ordinary
UBS Wealth Management Australia Ltd Buy 19,865 2,309 UBS Wealth Management Australia Ltd Buy 5,016 583 UBS Wealth Management Australia Ltd Buy 4,455 1,099	ar-14	UBS Wealth Management Australia Ltd	Buy	2,590	301	Ordinary
UBS Wealth Management Australia Ltd Buy 5,016 583 USS Wealth Management Australia Ltd Buy 9,455 1,099	ar-14	UBS Wealth Management Australia Ltd	Buy	19,865	2,309	Ordinary
UBS Wealth Management Australia Ltd Buy 9,455 1,099	ar-14	UBS Wealth Management Australia Ltd	Buy	5.016	583	Ordinary
		1 ACC Months & Annoncement Acceptalis 1 and	9	1117	3	
יין כלייני	21-Mar-14	UBS Wealth Management Australia Ltg	Buy	9.4551	200	Ordinary.

			7050	200	Crainary
21-Mar-14	UBS Securities Australia Ltd	3	4 115	(07/)	117
-Mar-14	UBS Securities Australia Ltd	Sell	3,810	(443)	Ordinary
23-Mar-14	UBS Securities Australia Itd	===	4 170	(204)	Craining
3-Mar-14	I IBS Securities Australia 14d	100	4,178	(407)	Ordinary
3.Mar.14	I IBC Countition Australia 14d	IIDC	557'5	(585)	Crainary
3-Mar-14	I IRS Securities Australia 14d	Bus	100 3	0 5	Courier
-Mar-14	LIBS Securities Australia Ind	Buck	197	100	Ordinary
3-Mar-14	UBS Securities Australia I to	Bix	14,000	1 640	Collinary
3-Mar-14	UBS Securities Australia Ltd	Sell	3 901	(456)	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buv	967	282	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Six	37 158	345	Cretinal
-Mar-14	URS Securities Australia I td	Bick	A 10F	717	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buv	3 266	382	Ordinary
3-Mar-14	UBS Securities Australia I td	Buy	1 154	135	Ordinary
3-Mar-14	UBS Securities Australia I to	Buy	11 628	1 260	Ordinary
3-Mar-14	URS Securities Australia Ltd	9	2 435	005,1	Ordinary
3-Mar-14	1 IRS Securities Australia 14d	100	1,555	(204)	Oldinary
-Mar.14	1/85 Securities Australia 14d	Test Seal	1120	(195)	Ordinary
3-Mar-14	UBS Securities Australia Ltd	3	1 964	(321)	Ordinary
3-Mar-14	LIBS Sequities Australia I to	3	513	(60)	Ordinary
3-Mar-14	I IRS Sequirities Australia 11d		10 151	1 100	Ordinary
3-Mar-14	LBS Securities Australia 1td	(F)	10,131	556	Countary
Mar-14	I IRC Constitute Australia 14d	900	106,1	233	Ordinary
3-Mar-14	I IRC Constitute Australia Ltd	And a	1,425	/91	Ordinary
2 Mar 14	UBS Secretion Australia ted	eny	0.038	761	Ordinary
D May 14	Opp Securities Australia Ltd	Buy	2,465	788	Ordinary
TING:	Obsideration Australia Ltd	Buy	3,003	352	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	8,018	940	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	6,398	750	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	265	70	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	222	56	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	13,571	1,591	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	2,508	294	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	1,979	232	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Sell	4,139	(483)	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Sell	4,134	(483)	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	1,180	138	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	3,788	443	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	316	37	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	6.293	736	Ordinary
3-Mar-14	UBS Securities Australia Ltd	Buy	9.174	1.073	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	20 221	2 365	Ordinary
3-Mar-14	UBS Securities Australia 11d	818	3770	400	Ordinary
3-Mar-14	UBS Securities Australia I to	Biry	2,720	308	Calman
3-Mar-14	UBS Securities Australia I to	Bin	7.075	100	Ordinary
2 Mar 14	I DO Countries Australia 140	, and	E/8'/	176	Ordinary
41-19W-C	Opp Securities Australia Ltd	Buy	070'1	071	Ordinary
Mar 14	LIDE Committee Australia Ltd	Sea.	1,042	771	Orginary
Nar-14	LIBC Counting Australia 14d	36	120'0	(0//)	Ordinary
A 1.62. 1.4	Libo AC Audrolin Broads	buy -3	177'06	660'6	Ordinary
Mar 17	The property of the property o	II AC	612,001	(21,102)	Ordinary
24-INIGI-14	UBS AC Australia Branch	Buy	855,19	10,/16	Ordinary
24-Wal-14	UBS AG AUStraila Branch	Buy	102,585	12,000	Ordinary
- INIGIL I #	UDD AG Australia branich	Buy	739,365	78,000	Ordinary
4-Mar-14	UBS AG London Branch	Buy	2,106,048	246,498	Ordinary
24-Mar-14	UBS AG London Branch	Sell	2,107,090	(246,498)	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Stock borrowed	NVA	400,000	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	33,774	3,951	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Seli	10,263	(1,200)	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	943,985	110,502	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Sell	691,553	(80,925)	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	25,590	3,000	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	214,172	25,059	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Sell	147.759	(17.261)	Ordinary
-Mar-14	UBS Securities Australia Ltd	Şeşl	2.107.090	(246.498)	vecipan
24 kAn 14					5
4	II IRS Societies Australia Ltd	Coll	200 674	(24 OE9)	Ordinary

+1-inlet-17	Opp securities Australia LIU	louy	32,449	3,747	Ordinary
1-Mar-14	I IBS Securities Australia 14d	, n 'Q	000 100	10.02	
21-Mar-14	UBS Securities Australia Ltd	Sell	67.531	(7 798)	Ordinary
1-Mar-14	UBS Securities Australia Ltd	Sell	448	(52)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	las	802	(Eb)	Ordinary
1-Mar-14	UBS Securities Australia I td	وا	1 576	(183)	Ordinary
1-Mar-14	I BS Securities Australia I td	100	3 1 46	(301)	Ordinaly
21-Mar-14	I IBC Countries Australia 14d	100	5115	(555)	O'Girany
21-Mar-14	I IRC Countries Australia 14d	liec d	3,156	(205)	Ordinary
:	COO COOLUMN CO	100	001,0	(/00)	Orainary
21-Mar-14	UBS Securities Australia Ltd	Seil	4,102	(477)	Ordinary
I-Mar-14	UBS Securities Australia Ltd	Seli	619	(72)	Ordinary
ar-14	UBS Securities Australia Ltd	Sell	7,745	(006)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	3,629	(422)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	1100	(124)	Ordinary
21-Mar.14	I BC Congrities Australia Ltd	log	1 000	(110)	Continue
4 4 4 au 1 4	I DO Commission Assessed Little	11-12	520,1	(611)	Guillaiy
<u>+</u> ;	ODS SECUTIVES AUSTRAIR LIG	Ž	3,182	(3/0)	Ordinary
-Mar-14	UBS Securities Australia Ltd	Set	1,944	(525)	Ordinary
ar-14	UBS Securities Australia Ltd	Sell	4,288	(438)	Ordinary
1-Mar-14	UBS Securities Australia Ltd	Sell	3,737	(435)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	2,027	(236)	Ordinary
1-Mar-14	UBS Securities Australia Ltd	Sell	430	(20)	Ordinary
ar-14	UBS Securities Australia Ltd	Se	301	(32)	Ordinary
21-Mar-14	ILIBS Securities Australia I to	i d	3,668	(90)	Ordinan
1,1	I ID Committee Australia Led	1123	4 702	(071)	O CHINGIN
1	Obs securities Australia Ltu	100	4,702	(acc)	Ordinary
2 I-IVIGIT- 14	UBS Securities Australia Ltd	Sell	4,633	(889)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	3,859	(449)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	3,431	(338)	Ordinary
1-Mar-14	UBS Securities Australia Ltd	Sell	3,560	(414)	Ordinary
ar-14	UBS Securities Australia Ltd	-Se	3.017	(351)	Ordinary
ar-14	I BS Securities Australia tito	ll d'S	1 800	(1221)	Croinso
-Mar-14	I IBS Securities Australia 14d	Coll	2663	010	Ordinary
	COO Secon ties South and Eric	191	2,003	(016)	Cicilialy
21-Mar-14	UBS Securities Australia Ltd	Self	550	(S	Ordinary
1-Mar-14	UBS Securities Australia Ltd	Sell	829	(100)	Ordinary
ar-14	UBS Securities Australia Ltd	Sell	332	(33)	Ordinary
ar-14	UBS Securities Australia Ltd	Sell	661	(77)	Ordinary
1-Mar-14	11.185 Securities Australia I td	رواا	43	(3)	Ordinan
21-Mar-14	I IRC Countries Australia 14d	3	1 750	(500)	Ordinary
	Principle (1997) 200 (1997)		20 (1)	(+02)	Countary
1	Obsideral lites Australia Ltd	Sell	/0/1	(661)	Ordinary
1-Mar-14	UBS Securities Australia Ltd	Zeit.	17	(2)	Ordinary
ar-14	UBS Securities Australia Ltd	Seil	137	(16)	Ordinary
-Mar-14	UBS Securities Australia Ltd	Sell	274	(32)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	1.594	(186)	Ordinary
21-Mar-14	UBS Securities Australia Itd	llaS	3 531	(412)	Ordinan
	Tibe Committee Australia 14d	1100	1000	(7) (1)	Cidinaly
-Mar- : 4	UBS Securities Australia Ltd	Sell	//	6)	Ordinary
1-Mar-14	UBS Securities Australia Ltd	Seli	3,037	(354)	Ordinary
ar-14	UBS Securities Australia Ltd	Sell	3,479	(405)	Ordinary
1-Mar-14	UBS Securities Australia Ltd	les	2.640	(307)	Ordinary
1-Mar-14	LIBS Securities Australia Ltd	197	2 804	(308)) decipies
21 84 22	1105 Commission Assessed and	130	2,904	(320)	Godinary
:	Obs securities Australia Ltd	- N	047'	(145)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	3,054	(326)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	1,356	(158)	Ordinary
ar-14	UBS Securities Australia Ltd	Sell	3.878	(452)	Ordinary
21-Mar-14	I IRS Securities Australia Ltd	lla?	154	(10)	Ordinan
	100 C	===	10-	(01)	o duli ai y
-IVIGIT-14	Obs securities Australia Ltd	Sell	4,647	(541)	Ordinary
-Mar-14	UBS Securities Australia Ltd	Sell	4,243	(494)	Ordinary
1-Mar-14	UBS Securities Australia Ltd	Sell	3,461	(402)	Ordinary
ır-14	UBS Securities Australia Ltd	Sell	439	(51)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Se	2 898	(337)	Ordinary
21-Mar-14	I BC Contribio Australia (14	Ties.	266	(100)	Ordinary
	The second of th	100	867	2	Cidinaly
21-Mar-14	UBS Securities Australia Ltd	sell	6	ĵ.	Ordinary
-Mar-14	UBS Securities Australia Ltd	Self	326	(38)	Ordinary
21-Mar-14	UBS Securities Australia Ltd	Sell	2,986	(348)	Ordinary
-Mar-14	UBS Securities Australia Ltd	Sell	103	(12)	Ordinary
-Mar-14	I IBS Societities Australia I to	3	201 c	(120)	Ordinan
	The Countries Activities and activities ac	100	501,5	(100)	Clairary
1	Obs securities Australia Ltd	Sell	2,233	(797)	Crainary
21-Mar-14	IRC Contration Assets 144	=			
	מכים ונובי שתיוו מום דות	Sei	1,804	(210)	Ordinary

24-Mar-14	ILIBS Societies Australia Ltd	190	1000	1000	
Mar-14	UBS Securities Australia Ltd	105	1170	(1942)	Ordinary
24-Mar-14	UBS Securities Australia 11d	<u> </u>	P95	(35)	Ordinary
24 Mar. 14	I IBC Constitution Australia Library	lles	2004	(00)	Ordinary
54-INIGI-14	Ubs securities Australia Ltd	nes.	2,203	(258)	Ordinary
54-IMI-14	UBS Securities Australia Ltd	Nell Nell	2,603	(302)	Ordinary
4 - IPINI-#7	Obs Securities Australia Ltd	nec .	2,845	(333)	Ordinary
Mar-14	UBS Securities Australia Ltd	Sell	4,583	(536)	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Sell	3,615	(423)	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Sell	1,924	(225)	Ordinary
24-Mar-14	Securities	Buy	27,557	3,223	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	6,515	762	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	669	81	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	2,600	655	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	6,840	800	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	9.644	1.128	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buv	39.663	4 639	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buv	864	101	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	5.985	700	Ordinary
Mar-14	UBS Securities Australia Ltd	Buv	34 311	4 013	Ordinan
24-Mar-14	UBS Securities Australia Ltd	Buv	86	01	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	879	20	Ordinary
4-Mar-14	UBS Securities Australia (1d	×18	282	CC	ording.
4-Mar-14	UBS Securities Australia Ltd	Birv	1 898	555	Crdinary
Mar-14	UBS Securities Australia I to	Buy	10 290	1 267	Coding
-Mar-14	Securities	Buy	80	1,502	0
24-Mar-14	URS Securities Australia Ind	Buv	305	100	Clumary
4-Mar-14	I IRS Securities Australia 14d	Billy	37.6	***	Clumary
A-Mar-14	LIBC Securities Australia 14d	, na	9/5	4 6	Ordinary
A Mar 14	Libe comittee Australia Ltd	ind d	pc	8	Orginary
A Mar 14	Upo Securities Australia Ltd	Buy	9/9′)	96	Ordinary
MINIGE I	Ops securities Australia Lid	and	950,61	1,/59	Ordinary
Mal-14	Obs Securities Australia Ltd	Buy	3,027	354	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	3,121	365	Ordinary
Mar 14	UBS Securities Australia Ltd	Buy	14,475	1,693	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	222	56	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	16,826	1,968	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	11,166	1,306	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	1,060	124	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	37,543	4,391	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	2,505	293	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	3,001	351	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	2.214	259	Ordinary
Mar-14	UBS Securities Australia Ltd	Buv	4 275	2005	Ordinary
Mar-14	I IRS Securities Australia I to	Busy	000	200	ordinal y
24-Mar-14	UBS Securities Australia Ltd	Buy	5003	1 10	ordinal y
24-Mar-14	I BC Countries Australia 1+d	Bury	200,6	200	Orginary
t - IPIN	Obs securities Australia Ltd	Buy	5,823	189	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	6,259	732	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	4,694	549	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	1,137	133	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	650	9/	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	4,711	551	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	4,284	501	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	32,268	3,774	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Seil	4,759	(256)	Ordinary
4-Mar-14	UBS Securities Australia Ltd	lleS	111	(61)	Ordinan
Mar-14	I IRS Securities Australia I to	3	900	(090)	Colingo
4-Mar-14	UBS Securities Australia Ltd	200	1 496	175	Ordinary
4-Mar-14	I IBS Securities Australia I to	28.2	1 623	201	Ordinary Ordinary
24-Mar-14	LIRS Securities Australia 14d	2.0	803 66	16.0	Oldinal y
A Mar 14	I IDC Cognition Australia Ltd	600	22,230	2,043	Ordinary
ti-lei	Obs Securities Australia Liu	Buy	393	46	Ordinary
24-IMar-14	UBS Securities Australia Ltd	Buy	26,214	3,066	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	2,214	259	Ordinary
4-Mar-14	UBS Securities Australia Ltd	Buy	17,767	2,078	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	4,933	577	Ordinary
24-Mar-14	UBS Securities Australia Ltd	Buy	2 420	coc	Ordinan
			1	607	5 5 5
24-Mar-14	UBS Securities Australia Ltd	Buv	2,086	244	Ordinary

	4,746	(222)	Ordinary
	644	(9/)	Ordinary
	4,164	(491)	Ordinary
	22,967	2,718	Ordinary
Securities Australia Ltd	306	(36)	Ordinary
UBS Securities Australia Ltd	308	(36)	Ordinary
	4,825	(695)	Ordinary
	10,254	(1,212)	Ordinary
UBS Securities Australia Ltd Buy	50,984	5,963	Ordinary
	30,327	3,547	Ordinary
ralia I td	111,612	13 054	Ordinary
UBS Securities Australia Ltd Sell	49.7	(51)	Ordinary
	761	(68)	Ordinary
	1,590	(186)	Ordinary
JBS Securities Australia Ltd Self	744	(83)	Ordinary
	137	(16)	Ordinary
	2,428	(284)	Ordinary
	6,361	(744)	Ordinary
lia Ltd Sell	12,681	(1,484)	Ordinary
	185	(89)	Ordinary
JBS Securities Australia Ltd	199,8	(468)	Ordinary
	17	(2)	Ordinary
JBS Securities Australia Ltd	1,486	(174)	Ordinary
	282	(33)	Ordinary
UBS Securities Australia Ltd	4,979	(283)	Ordinary
	6	1	Ordinary
	6,115	(716)	Ordinary
	17	(2)	Ordinary
UBS Securities Australia Ltd	3,361	(394)	Ordinary
	119	(14)	Ordinary
a Ltd Sell	512	(09)	Ordinary
	3,864	(453)	Ordinary
Obsideration of the control of the c	871	(41)	Ordinary
	010,8	(755)	Ordinary
USS Securities Australia Ltd	949	(LL)	Ordinary
	6///6	(447)	Ordinary
	3,510	(4)	Ordinary
UBS Securities Australia Ltd	950'1	(124)	Ordinary
	6	Ê	Ordinary
lia Ltd	26	e i	Ordinary
UBS Securities Australia Ltd	2,991	(351)	Ordinary
	3,651	(428)	Ordinary
	495	(28)	Ordinary
UBS Securities Australia Ltd Self	2,175	(255)	Ordinary
	572	(67)	Ordinary
	324	(38)	Ordinary
UBS Securities Australia Ltd	σ	(1)	Ordinary
ia Ltd Sell	2,448	(287)	Ordinary
JBS Securities Australia Ltd	102	(12)	Ordinary
JBS Securities Australia Ltd	2,280	(267)	Ordinary
JBS Securities Australia Ltd	128	(15)	Ordinary
UBS Securities Australia Ltd Sell	308	(36)	Ordinary
	855	(1001)	Ordinary
		(02/	Carried C
	666	(710)	Ordinary
	0.70,	(512)	O'GILIBIY
	#co,1	(717)	Ordinary
	80/1	(2007)	Ordinary
	3,446	(403)	Ordinary
	955	(42)	Ordinary
UBS Securities Australia Ltd	2,442	(286)	Ordinary
	444	(75)	Orginary
UBS Securities Australia Ltd Sell	623	(73)	Ordinary

	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	480	(3,707)	Ordinary
	ecurities Australia Ltd	رواز	31 631	(3,707)	Cignialy 1
 	Contract Account to			(10,10)	Charles and
	ACCURATION AND AND AND AND AND AND AND AND AND AN	Don't	120,16	1 101	Ordinary
	IDS Countries Australia Ltd	ind.	909'09	(102)	Ordinary
	Scotlines Australia too	and a	0,000	100	Ordinary
 	Jobs Securities Australia Ltd	gna	58,388	6,845	Ordinary
	JBS Securities Australia Ltd	Sell	443	(25)	Ordinary
 	JBS Securities Australia Ltd	Sell	800	(94)	Ordinary
7 2 2 7 7 7 	ecurities Australia Ltd	Seli	09	(2)	Ordinary
	JBS Securities Australia Ltd	Self	1,582	(186)	Ordinary
	JBS Securities Australia Ltd	Sell	3,074	(361)	Ordinary
	JBS Securities Australia Ltd	Sell	477	(95)	Ordinary
	ecurities Australia Ltd	ll as	σ	()	Ordinary
	IRS Securities Australia 144	Coll	5 220	(02.3)	000
T	IDS Societion American and	100	0//6	(0/0)	Ordinary
200	eculines Adstraila Ltd	120	766,6	(00/)	Ordinary
	Securities Australia Ltd	Sell	1,821	(214)	Ordinary
Ť	UBS Securities Australia Ltd	llec	4,655	(547)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	1,325	(156)	Ordinary
- 1	ecurties Australia Ltd	Sell	699	(28)	Ordinary
- 1	ecurities Australia Ltd	Sell	1,096	(129)	Ordinary
5-Mar-14 UBS Se	UBS Securities Australia Ltd	Sell	1,036	(122)	Ordinary
5-Mar-14 UBS Se	UBS Securities Australia Ltd	Sell	731	(98)	Ordinary
Г	UBS Securities Australia I td	3	1019	(120)	Ordinan
T	IBS Securities Australia I tol	Sell	PVU	(4.2)	(in in in
T	Countries Assessed and	18.5	1000	(L41)	Collinary
Ť	Upo Committees Australia Liu	E C	4,044	(547)	Orginary
7	ecurities Australia Ltd	- N	0/6/2	(349)	Ordinary
7	UBS Securities Australia Ltd	Sell Sell	604	(11)	Ordinary
5-Mar-14 UBS Se	UBS Securities Australia Ltd	Sell	1,345	(158)	Ordinary
5-Mar-14 UBS Se	UBS Securities Australia Ltd	Jac.	1 072	(126)	Ordinary
5-Mar-14 I IRS Se	Politities Australia I tol	9	809 /	(000)	Columbia
1	on white Australia I ad		000	(200)	O'GII (a)
2 cao + 1 - lair - C	Gos Securities Australia Ltu	Į,	90	(8)	Ordinary
-1	ecurities Australia Ltd	Sell	4,361	(513)	Ordinary
1	UBS Securities Australia Ltd	Seli	3,392	(400)	Ordinary
	ecurities Australia Ltd	Sell	704	(83)	Ordinary
5-Mar-14 UBS Se	UBS Securities Australia Ltd	Seli	3,521	(415)	Ordinary
5-Mar-14 UBS Se	UBS Securities Australia Ltd	Seli	3,515	(414)	Ordinary
5-Mar-14 UBS Se	ecurities Australia Ltd	Sell	773	(10)	Ordinary
ı	1 BC Securities Australia Ltd	3	2000	(444)	
1	the contract of the contract o	1130	464,0	1	Crusialy
П	eculices Australia Ltd	iac i	neg's	(477)	Ordinary
7	JBS Securities Australia Ltd	Sell	1,284	(151)	Ordinary
_	JBS Securities Australia Ltd	Sell	1,191	(140)	Ordinary
5-Mar-14 UBS Se	UBS Securities Australia Ltd	Seil	2.441	(787)	Ordinary
1	Pourities Australia I to	3	EV8 1	(158)	Grain
1	I IRS Securities Australia I to	9	100	(90)	ording)
П	Continuo Australia I tal		177	(02)	Cidillary
ı	ecurines Australia Liu	Jac II	810	(a)	Ordinary
	UBS Securities Australia Ltd	Sell	145	(17)	Ordinary
	UBS Securities Australia Ltd	Sell	170	(20)	Ordinary
	ecurities Australia Ltd	Sell	536	(63)	Ordinary
5-Mar-14 UBS Se	UBS Securities Australia Ltd	Sell	1.054	(124)	Ordinary
T	IBS Sequities Australia I to	Soll	476	(20)	in in it
Ť	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -		0	(or)	Geraly
2-Mar-14 UB3 Se	ecurities Australia Ltd	Seil	178	(15)	Ordinary
	UBS Securities Australia Ltd	Seil	928	(109)	Ordinary
	ecurities Australia Ltd	Sell	919	(108)	Ordinary
5-Mar-14 UBS Se	JBS Securities Australia Ltd	Sell	1.969	(1)31)	Ordinary
Ť	IRS Securities Australia I to	les	1 896	(203)	Ordinan
5-Mar-14 (1RS Sp.	IRS Securities Australia I td	Sell	3 262	(204)	Croiner
T	DO SCOTTON A THE PER		202,0	(+00)	Cidinaly
T	curines Australia Ltd	Self	2,0/3	(244)	Ordinary
7	UBS Securities Australia Ltd	Sell	732	(86)	Ordinary
25-Mar-14 UBS Se	UBS Securities Australia Ltd	Sell	1,703	(200)	Ordinary
_	ecurities Australia Ltd	Sell	1,549	(182)	Ordinary
Τ		9	1 140	(134)	Ordinan
T		130	900 6	(1010)	100
Ť	SCUTTIES Australia Ltu	llac .	3,008	(353)	Ordinary
┪	JBS Securities Australia Ltd	Sell	1,600	(188)	Ordinary
25-Mar-14 UBS Sev	JBS Securities Australia Ltd	Sell	086	(115)	Ordinary
П	ecurities Australia Ltd	Sell	2.604	(306)	Ordinary
Т	COO Securities Australia Etc	300	2,004	(000)	Ordinary

Buy 30,806 3 Buy 12,919 1,1 Buy 12,919 1,1 Buy 1,130 1,1 Buy 1,1004 1,1 Sell 1,1004 1,1 Sell 1,1006 1,1 Sell 1,	The state of the s				O.dinas
Buy 12,919 19,000 19,000 19,000 10,0	utilies Australia Ltd	hod	2,993	350	Oldinary
Buy 1,2919 Buy	curities Australia Ltd	Buy	30,806	3,603	Ordinary
Buy \$5.976 Buy \$5.976 Buy \$5.976 Buy \$3.541 Buy \$3.146 Buy \$7.139 Buy \$7.139 Buy \$7.130 Buy \$7.150 Buy \$7.150 Buy \$7.150 Buy \$7.152 Buy \$7.152 Buy \$7.153 Buy \$7.153 Buy \$7.153 Buy \$7.153 Buy \$7.153 Buy \$7.153 Buy \$7.103 Buy \$7.103 Buy \$7.103 Buy \$7.103 Buy \$7.103 Buy \$7.103 Buy \$7.206 Sell \$7.442 Sell \$7.442 Sell \$7.442 Sell \$7.242 Sell \$7.242 Sell \$7.242	curities Australia Ltd	Buy	12,919	1,511	Ordinary
Buy 3,591	curities Australia Ltd	Buy	5,976	669	Ordinary
Buy 7,139	curities Australia Ltd	Buy	3,591	420	Ordinary
Buy 3,146 Buy 1,000 Buy 1,000 Buy 1,163 Buy 1,164 Buy 1,164 Buy 1,164 Buy 1,164 Buy 1,164 Buy 1,164 Buy 1,162 Buy 1,163 Buy 1,263 Buy 1,263 Buy 1,263 Buy 1,263 Sell 3,73 Sell 3,73 Sell 3,73 Sell 3,73 Sell 3,73 Sell 3,73 Sell 4,27 Sell 3,128 Sell 3,138 Se	ecurities Australia Ltd	Buy	7,139	835	Ordinary
Buy 3,146 Buy 1,000 Buy 21,580 Buy 2,1580 Buy 2,1580 Buy 2,1580 Buy 2,1657 Buy 2,1657 Buy 1,167 Buy 1,168 Buy 1,168 Buy 1,168 Buy 1,168 Buy 1,168 Buy 1,168 Buy 1,268 Sell 1,206 Sell 3,12 Sell	ecurities Australia Ltd	N Buy	3.146	368	Ordinary
Buy 1,000 Buy 1,000 Buy 1,163 Buy 1,163 Buy 1,163 Buy 1,163 Buy 1,163 Buy 1,161 Sell 3,171 Buy 1,161 Sell 3,122 Buy 1,161 Sell 3,122 Buy 1,161 Buy 1,1	ecurities Australia I td	Brv	3 146	368	Ordinary
Buy 1,004	ocurities Australia 14d	0.0	000	25:	ordinary Fig. 1
Sell	eculties Australia Ltd	ng	000,1	/=	Ordinary
Sell 3-98	ecurities Australia Ltd	Buy	71,580	2,524	Ordinary
Buy 14672	ecurities Australia Ltd	Sell	3,998	(467)	Ordinary
Buy 8/347 Buy 2,155 Buy 2,155 Buy 2,155 Buy 2,155 Buy 1,1064 Buy 1,1064 Buy 1,132 Buy 1,132 Buy 1,132 Buy 1,132 Buy 1,152 Buy 1,513 Buy 1,513 Buy 1,513 Buy 1,513 Seli 1,263 Seli 3,28 Seli 3,	ecurities Australia Ltd	Buy	14,672	1,716	Ordinary
Buy 1,163	ecurities Australia Ltd	Buv	8.747	1.023	Ordinary
Buy	ecurities Australia I td	Riv	1163	751	Ordinan
Buy	Complete American Company (c. 14-4)	62.0	2	200	Ordinary C
Buy 2,078	securities Australia Ltd	And	2,155	757	Orginary
Buy 1,054	securities Australia Ltd	Buy	2,078	243	Ordinary
Buy 11,064	Securities Australia Ltd	Buy	21,657	2,533	Ordinary
Buy 3,223	Securities Australia Ltd	Buy	11,064	1,294	Ordinary
Buy 3,223	Securities Australia Ltd	Buy	265	7	Ordinary
Buy 1,122	Coourities Australia 14d	200	2000	12.5	(in the contract of
Buy 1,472	Securities Australia Lite	and a	5,445	2//	Cidinaly
Buy 1,872	securities Australia Ltd	Buy	4,121	482	Ordinary
Buy 19,152 Sell 1,407 Sell 1,613 Buy 108,349 Stock borrowed 1,55,430 Buy 108,349 Buy 108,432 Sell 1,044,529 Sell 3,109 Sell 3,100 S	Securities Australia Ltd	Buy	1,872	219	Ordinary
Buy 19,152 Sell 4,802 Sell 1,613 Buy 25,809 Stock borrowed 15,421 Buy 25,809 Sell 1,045,529 Sell 2,61 Sell 3,719 Sell 3,719 Sell 3,719 Sell 4,521 Sell 3,719 Sell 3,719 Sell 3,719 Sell 3,719 Sell 3,710 Sell	Securities Australia Ltd	Buv	22.692	2.654	Ordinary
Sell	Securities Australia I to	Bin	10 157	2 240	Ordinary
Sell	Constitute Americal to	103	2007	7,577	Colonial y
Sell	accurred Australia Liu	2611	4,002	(100)	Orainary
Se 1,1613	securities Australia Ltd	Sell	1,407	(164)	Ordinary
Buy 108.349	Securities Australia Ltd	Sell	1,613	(188)	Ordinary
Buy 252,809 Buy Sicok borrowed 155,421 Sicok 256 Sicok 256	AG Australia Branch	Buv	108.349	12.719	Ordinary
Stock barrowed 15,421 15,421 15,421 15,421 15,421 15,421 15,421 15,421 15,421 15,432 15	AG Australia Branch	Biv	252 809	757 00	Ordinan
Self 155,421 156,421 156,421 156,421 156,421 156,421 156,422 156,423	Committee American Committee in the	500	202,000	7 200 000 C	Clumary
See 155,421	Pin Billings Verification	Stock Collowed	YA.	2,000,000	Ordinary
Buy 68,593	securities Australia Ltd	Sell	155,421	(18,293)	Ordinary
Buy 1,044,529 Self 1,044,529 Self 42,492 Self 5,46 Self 1,785 Self 3,608 Self 3,608 Self 3,608 Self 3,719 Self 3,719 Self 4,856 Self 4,856 Self 4,856 Self 4,856 Self 4,856 Self 4,427 Self 4,427 Self 4,427 Self 4,527 Self 3,12 Self 3,13 Self 3,13	securities Australia Ltd	Buy	68,593	8,073	Ordinary
Seii 1,044,529 Seii 4,492 Seii 1,044,529 Seii 1,046,529 Seii 1,046 Seii 2,028 Seii 3,028 Seii 3,028 Seii 3,028 Seii 4,427 Seii 4,427 Seii 4,427 Seii 4,427 Seii 4,541 Buy 683,645 Seii 3,028 Seii 1,067 Seii 4,541 Buy 683,645 Seii 3,028 Seii 3,028 Seii 1,067 Seii 3,028 Seii 3,028 Seii 1,067 Seii 3,028 Seii 4,028 Seii 3,028 Seii 4,028 Seii 3,028 Seii 4,028 Seii 4,028 Seii 5,028 Seii 5,028 Seii 6,028 Seii 6,028 Seii 6,028 Seii 7,028 Seii 7,028 Seii 8,028 Seii 8,028 Seii 9,028 Seii	Securities Australia Ltd	Buy	171,779	79,626	Ordinary
Seli	Securities Australia Ltd	Sell	1,044,529	(122,701)	Ordinary
1,785 1,785 1,785 1,785 1,785 1,881 1,88	Securities Australia Ltd	Seli	42,492	(4.994)	Ordinary
Sell S46 Sell 3.608 Sell 3.79 Sell 3.79 Sell 3.79 Sell 3.79 Sell 3.79 Sell 4.856 Sell 4.427 Sell 3.728 Sell 3.720 Sell 3.75 Sell 3.	Wealth Management Australia Ltd	Sell	1 785	(010)	Ordinary
Sell	Securities Australia i td	19	246	(64)	Ordinary
Sell 3,881 1,984 1,986	Cod intion Australia 14d	=======================================	200	(4)	Ordinary
Sell	The state of the s	130	/00	(C#)	Collinary
Sell 3,608 Sell 3,608 Sell 4,856 Sell 7,19 Sell 1,906 Sell 1,906 Sell 4,427 Sell 4,427 Sell 6,83,645 Sell 7,191 Sell 7,191 Sell 8,312 Sell 8,312 Sell 3,122 Sell 3,123 Sell 3,122 Sell 3,123 Sell 4,124 Sell 4,124 Sell 4,124 Sell 4,124 Sell 5,125 Sell 5,125 Sell 6,125 Sell 7,125 Sell 7,125 Sell 8,125 Se	Securities Australia Ltd	Sell	3,881	(455)	Ordinary
Sell 3,928 Sell 3,719 Sell 68 Sell 76 Sell 1,906 Sell 2,900 Sell 8,900 Sell 8,2,900 Sell 8,122 Sell 8,122 Sell 8,900 Sell 8,900 Sell 8,900 Sell 8,900 Sell 9,900 Sell 8,900 Sell 9,900 Sell 9,9	securities Australia Ltd	Sell	3,608	(423)	Ordinary
Sell 3,719 Sell 4,856 Sell 3,038 Sell 1,906 Sell 1,906 Sell 4,427 Sell 4,427 Sell 4,427 Buy 10,672 (1 Buy 683,645 80 Sell 3,75 (20 Sell 3,122 (3 Sell 1,59,170 (18 Sell 7,703 (1 Sell 7,703 (2 Sell 7,703 (1 Sell 3,537 (1 Sell 7,703 (1 Sell 3,50 (49 (1) Sell 9,537 (1 Sell 3,50 (1 3,50	Securities Australia Ltd	Sell	3,928	(460)	Ordinary
Sell	Securities Australia Ltd	Sel	3 719	(436)	Ordinary
Sell	Societies Australia I tot	Coll	9000	(023)	Continue
Sell Sell Sell Sell Sell Sell Sell Sell Sell 1,906 Sell 1,906 Sell 1,906 Sell 4,427 Sell 4,427 Sell 4,427 Sell 8,23,645 Sell Sell 3,122 Sell 3,123	Securities Australia Ltd	Į,	4,000	(0/c)	Orginary
Sell Sel Sel Sel Sel Sel Sel Sel Sel 1.906 Sel 1.906 Sel 1.906 Sel Sel	Securities Australia Ltd	Sell	56	(3)	Ordinary
Sell 3,038 1,006	Securities Australia Ltd	Sell	89	<u>(8)</u>	Ordinary
Self 1,906 Self 1,906 Self 4,427 Self 4,427 Self 6,83,645 Self 8,703 Self 8,703 Self 8,703 Self 8,703 Self 8,703 Self 8,94,94 Self 3,122 Self 3,123	Securities Australia Ltd	-Se	3.038	(357)	Ordinary
Sell	Societies Australia Ltd	ISS	2001	(100)	, deline
Sell	Securities Australia Lid	100	906,1	(224)	Claridiy
Sell	seculities Australia Liu	Oei:	6	(14)	Ordinary
Sell 4,541 Buy 77 Sell 10,672 Sell 230 Sell 375 Sell 3,122 Sell 15,312 Sell 15,312 Sell 15,317 Sell 7,703 Sell 9,494 Sell 3,637 Sell 3	securities Australia Ltd	Sell	4,427	(519)	Ordinary
Buy 77 78 78 79 79 79 79 79	securities Australia Ltd	Sell	4,541	(233)	Ordinary
Sell	securities Australia Ltd	Buy	77	6	Ordinary
Sell 230	Antities Australia 11d	وا	10.672	(1 2¢ 4)	Ordinan
Sell	100 miles		2/0/01	(407/1)	S Countries
Sell 230	eculines Australia Ltd	Buy	683,645	80, 45	Ordinary
Sel 375	ecurities Australia Ltd	Sell	230	(27)	Ordinary
Sell 82,920	ecurities Australia Ltd	Sell	375	(44)	Ordinary
Sell 3,122 Sell 19,170 (Sell 2,170 (Sell 3,170 (Sell	ecurities Australia Ltd	Sell	82.920	(9.721)	Ordinary
Sell 199,170 159,170	ion tritios Australia I td	Zell	3 122	(366)	Ordinan
Sell 35/178 13/	bt ellertain Agitin Co.		150 170	(10,550)	Ordinos
Sell 2,1038 Sell Sell Sell Sell Sell Sell State Sell Sell	The state of the s	in the second	071,601	(000'01)	Oldingly 0 '
Sel 7,703	ecurities Australia Ltd	Sell	869'17	(2,539)	Ordinary
Sell 9.494 Sall 9,537 Sell 350	ecurities Australia Ltd	Seil	7,703	(603)	Ordinary
Sell 9,537	ecurities Australia Ltd	Sell	9,494	(1,113)	Ordinary
CSC Sell	ecurities Australia Itd	les	9 537	(1 118)	Ordinary
7	Contract August Indian	100	2,557	(1,110)	Gullary
000	securities Australia LId	Sell	350	(41)	Ordinary
UBS Securities Australia Ltd Sell (securities Australia Ltd	Sell	77	(6)	Ordinary
207 AAF	ecurities Australia Ltd	وا	366 705	(066 67)	Ordinary
50/500			1011000	こくついし	5 5 5
	cultings Australia Ltd	- Se	21,649	(2.538)	Ordinary

1,000 1,00	26-Mar-14	I IBS Societion Ametrolia 14d	les	3.391	(410)	Ordinary
URS Securities Autorials Ltd		ספס סברתווונים אחזוומוומ דות				
USS Securities Australia Ltd Buy 1,539 4,64 USS Securities Australia Ltd Buy 6,830 8,55 USS Securities Australia Ltd Buy 6,830 8,55 USS Securities Australia Ltd Buy 7,20 1,70 USS Securities Australia Lt	26-Mar-14	UBS Securities Australia Ltd	Sell	6,827	(824)	Ordinary
USS Securities Australia Ltd Bluy S8 677 5651 USS Securities Australia Ltd Bluy S8 677 5651 USS Securities Australia Ltd Bluy S8 677 5708 USS Securities Australia Ltd Bluy S8 677 5709 35 USS Securities Australia Ltd Bluy S9 677 5700 USS Securities Australi	26-Mar-14	UBS Securities Australia Ltd	Buy	3,349	404	Ordinary
UNIS Securities Australia (LIC) Bluy S8.677 7.078 UNIS Securities Australia (LIC) Bluy 6.839 2.82 UNIS Securities Australia (LIC) Bluy 6.83 2.82 UNIS Securities Australia (LIC) Bluy 2.271 2.86 UNIS Securities Australia (LIC) Bluy 3.474 4.19 UNIS Securities Australia (LIC) Bluy 1.623 1.724 UNIS Securities Australia (LIC) Bluy 1.725 1.728 UNIS Securities Australia (LIC) Bluy 2.727 1.728 UNIS Securities Australia (LIC) Bluy 1.728 1.728 UNIS Securities Australia (LIC) Bluy 1.728 1.728 UNIS Securities Australia (LIC) Bluy 1.728 1.728 UNIS Securities Australia (LIC) Bluy 1.729 1.728	26-Mar-14	UBS Securities Australia Ltd	Buy	46,847	5,651	Ordinary
UBS Securities Australia Ltd Blyy 6.89 8.25 8.15 LBS Securities Australia Ltd Blyy 6.89 1.1 1.1 LBS Securities Australia Ltd Blyy 6.89 1.1 1.1 LBS Securities Australia Ltd Blyy 8.47.4 4.19 1.1 LBS Securities Australia Ltd Blyy 8.29.8 1.1 1.0 LBS Securities Australia Ltd Blyy 8.29.8 1.7 1.0 LBS Securities Australia Ltd Blyy 1.2.80 1.0 1.0 LBS Securities Australia Ltd Blyy 1.2.80 1.7 1.0 LBS Securities Australia Ltd Blyy 1.2.80 1.7 1.0 LBS Securities Australia Ltd Blyy 1.2.20 2.2 1.0 LBS Securities Australia Ltd Blyy 1.2.20 2.2 1.0 LBS Securities Australia Ltd Blyy 1.2.20 2.2 2.2 LBS Securities Australia Ltd Blyy 1.2.2 2.2 2.2 2.2 LBS Securities Au	26-Mar-14	UBS Securities Australia Ltd	Buy	58,677	7,078	Ordinary
USS Securities Australia Ltd Bluy 6.839 8.85 1.5 USS Securities Australia Ltd Bluy 1.47 1.7 1.7 USS Securities Australia Ltd Bluy 1.42 1.7 1.7 1.7 USS Securities Australia Ltd Bluy 1.42 1.7 1.7 1.7 USS Securities Australia Ltd Bluy 1.42 1.7	26-Mar-14	UBS Securities Australia Ltd	Buy	290	35	Ordinary
USS Securities Australia Ltd	26-Mar-14	UBS Securities Australia Ltd	Buy	6,839	825	Ordinary
UBS Securities Australia Ltd Buy 18 UBS Securities Australia Ltd Buy 141 17 UBS Securities Australia Ltd Buy 2,371 2,87 UBS Securities Australia Ltd Buy 3,474 1,478 UBS Securities Australia Ltd Buy 1,475 1,703 UBS Securities Australia Ltd Buy 1,475 1,703 UBS Securities Australia Ltd Buy 1,412 2,545 UBS Securities Australia Ltd Buy 1,412 2,548 UBS Securities Australia Ltd Buy 1,559 2,548 UBS Securities Australia Ltd Buy 1,559 2,549 UBS Securities Australia Ltd Buy 1,558 1,65 UBS Securities Australia Ltd Buy 1,558 1,158 UBS Securities Australia Ltd Buy 1,558 <td>26-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>8</td> <td>-</td> <td>Ordinary</td>	26-Mar-14	UBS Securities Australia Ltd	Buy	8	-	Ordinary
UBS Securities Australia Ltd Bluy 141 17 UBS Securities Australia Ltd Bluy 2,371 4.99 UBS Securities Australia Ltd Bluy 3,474 4.99 UBS Securities Australia Ltd Bluy 3,673 7.00 UBS Securities Australia Ltd Bluy 1,633 1,977 UBS Securities Australia Ltd Bluy 1,633 1,978 UBS Securities Australia Ltd Bluy 1,534 1,789 UBS Securities Australia Ltd Bluy 1,534 1,789 UBS Securities Australia Ltd Bluy 1,539 1,544 UBS Securities Australia Ltd Bluy 1,539 1,548 UBS Securities Australia Ltd Bluy 1,750 2,500 UBS Securities Australia Ltd Bluy 2,007 3,701 UBS Securities Australia Ltd	26-Mar-14	UBS Securities Australia Ltd	Buy	8	-	Ordinary
USS Securities Australia Ltd Bluy 2,371 2,86 USS Securities Australia Ltd Bluy 2,020 2,020 USS Securities Australia Ltd Bluy 1,125 1,000 USS Securities Australia Ltd Bluy 1,000 1,000 USS Securities Australia Ltd	26-Mar-14	UBS Securities Australia Ltd	Buy	141	171	Ordinary
UBS Securities Australia Ltd Bly 3,744 4.95 UBS Securities Australia Ltd Bly 5,020 7,000 UBS Securities Australia Ltd Bly 1,533 1,97 UBS Securities Australia Ltd Bly 1,521 2,506 UBS Securities Australia Ltd Bly 1,537 2,94 UBS Securities Australia Ltd Bly 1,539 2,544 UBS Securities Australia Ltd Bly 1,559 2,544 UBS Securities Australia Ltd Bly 1,559 2,546 UBS Securities Australia Ltd Bly 1,559 2,521 UBS Securities Australia Ltd Bly 1,559 1,759 UBS Securities Australia Ltd Bly 1,539 1,759 UBS Securities Australia Ltd Bly 1,539 2,500 UBS Securities Australia Ltd Bly 1,730 2,500 UBS Securities Australia Ltd Bly 1,730 2,500 UBS Securities Australia Ltd Bly 2,003 2,500 UBS Securities Australia Ltd	26-Mar-14	UBS Securities Australia Ltd	NI S	2 371	286	Ordinary
UBS Securities Australia Ltd Buy 5,037 7,030 UBS Securities Australia Ltd Buy 1,126 1,005 UBS Securities Australia Ltd Buy 1,126 1,005 UBS Securities Australia Ltd Buy 1,126 1,005 UBS Securities Australia Ltd Buy 2,15,23 2,546 UBS Securities Australia Ltd Buy 2,15,23 2,547 UBS Securities Australia Ltd Buy 2,15,23 2,548 UBS Securities Australia Ltd Buy 2,15,23 2,548 UBS Securities Australia Ltd Buy 1,2,307 1,528 UBS Securities Australia Ltd Buy 1,2,307 1,528 UBS Securities Australia Ltd Buy 1,2,27 1,528 UBS Securities Australia Ltd Buy 1,2,27 1,528 UBS Securities Australia Ltd Buy 2,0,27 2,2,29 UBS Securities Australia Ltd Buy 2,0,27 2,2,29 UBS Securities Australia Ltd Buy 2,0,27 2,2,27 UBS Securities Australi	26-Mar-14	I BC Counties Australia I to	(F)	2 474	3 5	Ordinary
UBS Securities Autorialia Ltd Buy 1,633 1,703 UBS Securities Autorialia Ltd Buy 1,633 1,97 UBS Securities Autorialia Ltd Buy 1,153 1,97 UBS Securities Autorialia Ltd Buy 1,27 1,244 UBS Securities Autorialia Ltd Buy 1,243 1,244 UBS Securities Autorialia Ltd Buy 1,251 1,244 UBS Securities Autorialia Ltd Buy 1,251 1,251 UBS Securities Autorialia Ltd Buy 4,250 2,50 UBS Securities Autorialia Ltd Buy 4,250 2,50 UBS Securities Autorialia Ltd Buy 4,250 2,50 UBS Securities Autorialia Ltd Buy 1,251 1,251 UBS Securities Autorialia Ltd Buy 1,252 1,25 UBS Securities Autorialia Ltd Buy 2,000 1,24 UBS Securities Autorialia Ltd Buy 2,000 1,24 UBS Securities Autorialia Ltd Buy 2,000 2,000 UBS Securities Autorialia Ltd	26-Mar. 14	I IRC Countries Australia (44)	600	4/4/0	4 1000	Oldillaly
UBS Securities Australia Ltd BMY 1,638 197 UBS Securities Australia Ltd BMY 1,639 1,703 UBS Securities Australia Ltd BMY 1,249 1,703 UBS Securities Australia Ltd BMY 1,249 1,759 UBS Securities Australia Ltd BMY 2,1821 2,524 UBS Securities Australia Ltd BMY 2,439 2,541 UBS Securities Australia Ltd BMY 2,439 2,541 UBS Securities Australia Ltd BMY 2,439 2,541 UBS Securities Australia Ltd BMY 2,550 2,000 UBS Securities Australia Ltd BMY 1,559 1,259 UBS Securities Australia Ltd BMY 2,500 2,000 UBS Securities Australia Ltd <t< td=""><td>25 1424 14</td><td>DO C</td><td>Buy</td><td>050,050</td><td>000'/</td><td>Ordinary</td></t<>	25 1424 14	DO C	Buy	050,050	000'/	Ordinary
UBS Securities Australia Ltd BMy 1,633 197 UBS Securities Australia Ltd BMy 1,623 2,596 UBS Securities Australia Ltd BMy 1,237 2,534 UBS Securities Australia Ltd BMy 1,237 2,534 UBS Securities Australia Ltd BMy 2,433 2,534 UBS Securities Australia Ltd BMy 2,433 2,531 UBS Securities Australia Ltd BMy 4,590 2,981 UBS Securities Australia Ltd BMy 4,590 2,581 UBS Securities Australia Ltd BMy 1,558 1,851 UBS Securities Australia Ltd BMy 1,558 1,851 UBS Securities Australia Ltd BMy 2,007 2,427 UBS Securities Australia Ltd BMy 2,007 2,428 UBS Securities Australia Ltd BMy 2,007 2,421 UBS Securities Australia Ltd BMy 2,007 2,527 UBS Securities Australia Ltd BMy 2,007 2,527 UBS Securities Australia Ltd <td< td=""><td>41-1PM-07</td><td>Obsisted Australia Ltd</td><td>Buy</td><td>80</td><td></td><td>Ordinary</td></td<>	41-1PM-07	Obsisted Australia Ltd	Buy	80		Ordinary
UBS Securities Autorials Ltd BMy 14,126 1,038 UBS Securities Autorials Ltd BMy 21,827 2,596 UBS Securities Autorials Ltd BMy 21,827 2,596 UBS Securities Autorials Ltd BMy 2,439 2,345 UBS Securities Autorials Ltd BMy 2,439 2,341 UBS Securities Autorials Ltd BMy 2,439 2,541 UBS Securities Autorials Ltd BMy 2,535 1,256 UBS Securities Autorials Ltd BMy 2,535 1,256 UBS Securities Autorials Ltd BMy 1,535 1,14 UBS Securities Autorials Ltd BMy 1,535 1,14 UBS Securities Autorials Ltd BMy 1,535 1,14 UBS Securities Autorials Ltd BMy 2,535 1,14 UBS Securities Autorials Ltd <t< td=""><td>20-Mar-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1,633</td><td>197</td><td>Ordinary</td></t<>	20-Mar-14	UBS Securities Australia Ltd	Buy	1,633	197	Ordinary
UBS Securities Australia Ltd Buy 1521 2,554 UBS Securities Australia Ltd Buy 1,241 2,554 2,543 2,544 UBS Securities Australia Ltd Buy 2,433 2,544 2,543 2,544 UBS Securities Australia Ltd Buy 2,433 2,531 1,525 UBS Securities Australia Ltd Buy 1,525 2,000 UBS Securities Australia Ltd Buy 4,560 2,527 2,627 UBS Securities Australia Ltd Buy 7,706 9,29 UBS Securities Australia Ltd Buy 2,603 7,73 UBS Securities Australia Ltd	26-Mar-14	UBS Securities Australia Ltd	Buy	14,126	1,703	Ordinary
UBS Securities Australia Ltd Buy 195.390 23.345 UBS Securities Australia Ltd Buy 1.2807 1.544 UBS Securities Australia Ltd Buy 2.433 2.534 UBS Securities Australia Ltd Buy 7.573 5.1 UBS Securities Australia Ltd Buy 7.573 5.8 UBS Securities Australia Ltd Buy 7.573 5.8 UBS Securities Australia Ltd Buy 7.575 1.526 UBS Securities Australia Ltd Buy 1.535 1.135 UBS Securities Australia Ltd Buy 1.535 1.135 UBS Securities Australia Ltd Buy 1.535 1.144 UBS Securities Australia Ltd Buy 1.535 1.144 UBS Securities Australia Ltd Buy 1.535 1.144 UBS Securities Australia Ltd Buy 5.007 2.007 2.425 UBS Securities Australia Ltd Buy 5.007 2.007 2.007 UBS Securities Australia Ltd Buy 5.007 2.007 2.007 <td>26-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>21,521</td> <td>2,596</td> <td>Ordinary</td>	26-Mar-14	UBS Securities Australia Ltd	Buy	21,521	2,596	Ordinary
UBS Securities Australia Ltd Buy 1,544 UBS Securities Australia Ltd Buy 2,433 2,544 UBS Securities Australia Ltd Buy 4,233 2,537 9,13 UBS Securities Australia Ltd Buy 1,539 1,539 1,530 <	26-Mar-14	UBS Securities Australia Ltd	Buy	193,530	23,345	Ordinary
UBS Securities Australia Ltd Bby 2,439 294 UBS Securities Australia Ltd Buy 1,531 51 UBS Securities Australia Ltd Buy 1,539 1,759 UBS Securities Australia Ltd Buy 1,539 1,759 UBS Securities Australia Ltd Buy 671 105 UBS Securities Australia Ltd Buy 6,70 1,83 UBS Securities Australia Ltd Buy 1,243 1,259 UBS Securities Australia Ltd Buy 1,341 2,22 UBS Securities Australia Ltd Buy 1,341 2,22 UBS Securities Australia Ltd Buy 1,375 1,58 UBS Securities Australia Ltd Buy 1,375 1,58 UBS Securities Australia Ltd Buy 2,007 2,42 UBS Securities Australia Ltd Buy 2,007 3,73 UBS Securities Australia Ltd Buy 2,323 3,64 UBS Securities Australia Ltd Buy 2,323 3,67 UBS Securities Australia Ltd Buy	26-Mar-14	UBS Securities Australia Ltd	Buv	12.807	1.544	Ordinary
URS Securities Australia Ltd Bay 7,573 913 UBS Securities Australia Ltd Buy 1,573 913 UBS Securities Australia Ltd Buy 1,589 1,500 UBS Securities Australia Ltd Buy 1,589 1,500 UBS Securities Australia Ltd Buy 4,560 5,88 UBS Securities Australia Ltd Buy 1,841 1,222 UBS Securities Australia Ltd Buy 1,841 1,222 UBS Securities Australia Ltd Buy 7,007 242 UBS Securities Australia Ltd Buy 7,007 242 UBS Securities Australia Ltd Buy 7,007 242 UBS Securities Australia Ltd Buy 5,599 754 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,599 755 UBS Securities Australia Ltd Buy 6,599 754 UBS Securities Australia Ltd Buy 6,599 754 UBS Securities Australia Ltd Buy <	26-Mar-14	UBS Securities Australia Ltd	Buv	2.439	294	Ordinary
UBS Securities Australia Ltd Buy 1,573 913 UBS Securities Australia Ltd Buy 1,4581 1,759 UBS Securities Australia Ltd Buy 1,6580 2,000 UBS Securities Australia Ltd Buy 1,258 1,258 UBS Securities Australia Ltd Buy 1,235 1,258 UBS Securities Australia Ltd Buy 1,237 1,6580 UBS Securities Australia Ltd Buy 1,237 1,658 UBS Securities Australia Ltd Buy 2,007 2,42 UBS Securities Australia Ltd Buy 2,007 2,42 UBS Securities Australia Ltd Buy 2,007 2,65 UBS Securities Australia Ltd Buy 2,007 2,65 UBS Securities Australia Ltd Buy 2,6734 3,701 UBS Securities Australia Ltd	26-Mar-14	UBS Securities Australia Ltd	Buy	423	-5	Ordinary
UBS Securities Australia Ltd Buy 14,551 1,759 UBS Securities Australia Ltd Buy 16,580 1,759 UBS Securities Australia Ltd Buy 4,260 5.000 UBS Securities Australia Ltd Buy 4,260 5.88 UBS Securities Australia Ltd Buy 1,341 2.22 UBS Securities Australia Ltd Buy 1,548 1,285 UBS Securities Australia Ltd Buy 7,706 9,834 UBS Securities Australia Ltd Buy 2,037 3,73 UBS Securities Australia Ltd Buy 2,629 7,74 UBS Securities Australia Ltd Bu	26-Mar-14	UBS Securities Australia I1d	Bit	7 573	013	Ordinary
UBS Securities Australia Ltd Buy 1,235 1,736 UBS Securities Australia Ltd Buy 1,6350 2,000 UBS Securities Australia Ltd Buy 1,6350 1,000 UBS Securities Australia Ltd Buy 1,535 1,835 UBS Securities Australia Ltd Buy 1,536 1,288 UBS Securities Australia Ltd Buy 1,518 1,288 UBS Securities Australia Ltd Buy 1,518 1,83 UBS Securities Australia Ltd Buy 1,518 1,83 UBS Securities Australia Ltd Buy 5,50 7,70 UBS Securities Australia Ltd Buy	26-Mar-14	I IRC Countries Americal and	6.0	14 504	010	Collinary
URS Securities Australia Ltd Buy F 580 L 500 URS Securities Australia Ltd Buy 6 96 1 585 URS Securities Australia Ltd Buy 4 360 588 URS Securities Australia Ltd Buy 1 341 222 URS Securities Australia Ltd Buy 1 341 222 URS Securities Australia Ltd Buy 1 341 222 URS Securities Australia Ltd Buy 1 341 1 228 URS Securities Australia Ltd Buy 1 518 1 51 URS Securities Australia Ltd Buy 2 000 7 74 URS Securities Australia Ltd Buy 5 000 7 74 URS Securities Australia Ltd Buy 2 000 7 74 URS Securities Australia Ltd Buy 2 000 7 75 URS Securities Australia Ltd Buy 2 000 7 75 URS Securities Australia Ltd Buy 2 000 7 75 URS Securities Australia Ltd Buy 2 000 7 75 URS Securities Australia Ltd Buy	26 142. 14	The control of the second of t	śng	14,031	60/1	Ordinary
UBS Securities Australia Ltd Buy 4,960 598 UBS Securities Australia Ltd Buy 1,535 188 UBS Securities Australia Ltd Buy 1,535 1,88 UBS Securities Australia Ltd Buy 1,536 1,88 UBS Securities Australia Ltd Buy 1,706 9,884 UBS Securities Australia Ltd Buy 1,706 9,894 UBS Securities Australia Ltd Buy 1,706 9,894 UBS Securities Australia Ltd Buy 1,518 1,81 UBS Securities Australia Ltd Buy 5,59 1,74 UBS Securities Australia Ltd Buy 5,59 7,7 UBS Securities Australia Ltd Buy 5,59 7,7 UBS Securities Australia Ltd Buy 5,59 7,7 UBS Securities Australia Ltd Buy 5,59 7,74 UBS Securities Australia Ltd Buy 5,59 7,74 UBS Securities Australia Ltd Buy 5,59 7,74 UBS Securities Australia Ltd Buy	20 Mar 14	Obsideration Australia Ltd	Buy	16,580	2,000	Ordinary
UBS Securities Australia Ltd Buy 4,960 558 UBS Securities Australia Ltd Buy 1,535 1885 UBS Securities Australia Ltd Buy 1,537 1,535 UBS Securities Australia Ltd Buy 1,706 929 UBS Securities Australia Ltd Buy 1,518 114 UBS Securities Australia Ltd Buy 1,518 114 UBS Securities Australia Ltd Buy 5,599 724 UBS Securities Australia Ltd Buy 5,599 735 UBS Securities Australia Ltd Buy 5,599 736 UBS Securities Australia Ltd Buy 5,599 737 UBS Securities Australia Ltd Buy 5,599 737 UBS Securities Australia Ltd Buy 5,599 737 UBS Securities Australia Ltd Buy 5,5	*1-JPINI-07	Obsisecurities Australia Ltd	Buy	8/1	105	Ordinary
UBS Securities Australia Ltd Buy 1,535 185 UBS Securities Australia Ltd Buy 1,535 185 UBS Securities Australia Ltd Buy 1,043 1,222 UBS Securities Australia Ltd Buy 7,06 293 UBS Securities Australia Ltd Buy 7,06 242 UBS Securities Australia Ltd Buy 7,07 242 UBS Securities Australia Ltd Buy 5,58 67 UBS Securities Australia Ltd Buy 6,09 77 UBS Securities Australia Ltd Buy 6,099 77 UBS Securities Australia Ltd Buy 6,599 794 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 6,599 796 UBS Securities Australia Ltd Buy 6,599 796 UBS Securities Australia Ltd Buy 6,599 796 UBS Securities Australia Ltd Buy 6,599 734 UBS Securities Australia Ltd Buy 6,599	20-Mar-14	UBS Securities Australia Ltd	Buy	4,960	298	Ordinary
UBS Securities Australia Ltd Buy 1,28B UBS Securities Australia Ltd Buy 1,28B UBS Securities Australia Ltd Buy 1,327 1,28B UBS Securities Australia Ltd Buy 1,706 293 UBS Securities Australia Ltd Buy 2,007 1,60 UBS Securities Australia Ltd Buy 6,009 7724 UBS Securities Australia Ltd Buy 6,009 7724 UBS Securities Australia Ltd Buy 6,009 7724 UBS Securities Australia Ltd Buy 6,509 795 UBS Securities Australia Ltd Buy 6,509 795 UBS Securities Australia Ltd Buy 6,509 724 UBS Securities Australia Ltd Buy 6,509 736 UBS Securities Australia Ltd Buy 6,509 736	25-Mar-14	UBS Securities Australia Ltd	Buy	1,535	185	Ordinary
UBS Securities Australia Ltd Buy 1,841 222 UBS Securities Australia Ltd Buy 81,739 9,824 UBS Securities Australia Ltd Buy 7,706 9,29 UBS Securities Australia Ltd Buy 7,706 9,20 UBS Securities Australia Ltd Buy 2,007 242 UBS Securities Australia Ltd Buy 5,55 67 UBS Securities Australia Ltd Buy 5,009 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 6,599 736 UBS Securities Australia Ltd Buy 6,599 737 UBS Securities Australia Ltd Buy 6,599 736 UBS Securities Australia Ltd Buy 6,599 737 UBS Securities Australia Ltd Buy 6,5	26-Mar-14	UBS Securities Australia Ltd	Виу	10,435	1,258	Ordinary
UBS Securities Australia Ltd Buy 81,739 9,854 UBS Securities Australia Ltd Buy 1,706 9,29 UBS Securities Australia Ltd Buy 2,007 242 UBS Securities Australia Ltd Buy 2,007 242 UBS Securities Australia Ltd Buy 555 673 UBS Securities Australia Ltd Buy 5,59 774 UBS Securities Australia Ltd Buy 2,073 3,211 UBS Securities Australia Ltd Buy 2,599 774 UBS Securities Australia Ltd Buy 2,573 3,211 UBS Securities Australia Ltd Buy 2,573 3,211 UBS Securities Australia Ltd Buy 2,573 3,28 UBS Securities Australia Ltd Buy 2,574 3,71 UBS Securities Australia Ltd Buy 5,574 3,70 UBS Securities Australia Ltd Buy 5,57 3,70 UBS Securities Australia Ltd Buy 6,595 7,98 UBS Securities Australia Ltd Buy	26-Mar-14	UBS Securities Australia Ltd	Buy	1,841	222	Ordinary
UBS Securities Australia Ltd Buy 7,706 929 UBS Securities Australia Ltd Buy 1,327 160 UBS Securities Australia Ltd Buy 2,007 242 UBS Securities Australia Ltd Buy 6,009 7724 UBS Securities Australia Ltd Buy 6,009 7724 UBS Securities Australia Ltd Buy 6,599 7724 UBS Securities Australia Ltd Buy 2,073 370 UBS Securities Australia Ltd Buy 2,599 735 UBS Securities Australia Ltd Buy 2,599 736 UBS Securities Australia Ltd Buy 2,599 736 UBS Securities Australia Ltd Buy 5,599 737 UBS Securities Australia Ltd Buy 5,599 737 UBS Securities Australia Ltd Buy 6,599 737 UBS Securities Australia Ltd Buy 6,599 737 UBS Securities Australia Ltd Buy 6,599 737 UBS Securities Australia Ltd Buy 6,5	26-Mar-14	UBS Securities Australia Ltd	Buy	81,739	9.854	Ordinary
UBS Securities Australia Ltd Buy 1,327 150 UBS Securities Australia Ltd Buy 1,327 152 UBS Securities Australia Ltd Buy 555 67 UBS Securities Australia Ltd Buy 6,009 77 UBS Securities Australia Ltd Buy 6,009 77 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 6,599 796 UBS Securities Australia Ltd Buy 6,599 796 UBS Securities Australia Ltd Buy 6,599 774 UBS Securities Australia Ltd Buy 6,599 736 UBS Securities Australia Ltd Buy 6,599 6,599 UBS Securities Australia Ltd Buy 6,599 6,399 UBS Securities Australia Ltd Buy 6,009 </td <td>26-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>7 705</td> <td>926</td> <td>Ordinary</td>	26-Mar-14	UBS Securities Australia Ltd	Buv	7 705	926	Ordinary
UBS Securities Australia Ltd Buy 1,027 1,02 UBS Securities Australia Ltd Buy 1,021 1,03 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 5,599 795 UBS Securities Australia Ltd Buy 6,599 724 UBS Securities Australia Ltd Buy 6,599 734 UBS Securities Australia Ltd Buy 6,599 735 UBS Securities Australia Ltd Buy 6,59	26-Mar-14	1 185 Segurities Australia 11d	B. I.	1 227	031	, and a
UBS Securities Australia Ltd Buy 1,518 1,52 UBS Securities Australia Ltd Buy 555 67 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 6,591 795 UBS Securities Australia Ltd Buy 5,673 3,701 370 UBS Securities Australia Ltd Buy 5,699 755 UBS Securities Australia Ltd Buy 1,820 2,388 UBS Securities Australia Ltd Buy 1,820 2,388 UBS Securities Australia Ltd Buy 6,595 754 UBS Securities Australia Ltd Buy 6,009 7,74 UBS Securities Australia Ltd Buy 6,009 7,24 UBS Securities Australia Ltd Buy 6,009 7,35 UBS Securities Australia Ltd <td< td=""><td>26-Mar-14</td><td>I IRS Societies Australia 14d</td><td>500</td><td>7200</td><td>200</td><td>Ciditaly</td></td<>	26-Mar-14	I IRS Societies Australia 14d	500	7200	200	Ciditaly
UBS Securities Australia Ltd Buy 15.18 183 UBS Securities Australia Ltd Buy 55.5 67.7 UBS Securities Australia Ltd Buy 6.099 724 UBS Securities Australia Ltd Buy 9.46 11.44 UBS Securities Australia Ltd Buy 26,599 755 UBS Securities Australia Ltd Buy 26,734 3.201 UBS Securities Australia Ltd Buy 26,734 3.221 UBS Securities Australia Ltd Buy 26,596 756 UBS Securities Australia Ltd Buy 26,596 758 UBS Securities Australia Ltd Buy 5,711 370 UBS Securities Australia Ltd Buy 6,599 724 UBS Securities Australia Ltd Buy 6,509 724 UBS Securities Australia Ltd Buy		CO Secondora Australia Etc	po)	7,00,	747	Ordinary
UBS Securities Australia Ltd Buy 635 67 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 2,332 2,599 UBS Securities Australia Ltd Buy 2,373 370 UBS Securities Australia Ltd Buy 2,5734 3,271 UBS Securities Australia Ltd Buy 2,5734 3,731 UBS Securities Australia Ltd Buy 2,573 3,736 UBS Securities Australia Ltd Buy 5,595 7,37 UBS Securities Australia Ltd Buy 6,595 7,37 UBS Securities Australia Ltd Buy 6,009 7,34 UBS Securities Australia Ltd Buy 6,009 7,34 UBS Securities Australia Ltd Buy	20-Mar-14	UBS Securities Australia Ltd	Buy	1,518	183	Ordinary
UBS Securities Australia Ltd Buy 6.08 774 UBS Securities Australia Ltd Buy 6.09 724 UBS Securities Australia Ltd Buy 6.59 795 UBS Securities Australia Ltd Buy 6.59 795 UBS Securities Australia Ltd Buy 6.59 795 UBS Securities Australia Ltd Buy 8.73 3.07 3.07 UBS Securities Australia Ltd Buy 6.595 795 UBS Securities Australia Ltd Buy 8.75 3.07 UBS Securities Australia Ltd Buy 6.595 736 UBS Securities Australia Ltd Buy 6.595 736 UBS Securities Australia Ltd Buy 6.099 7.24 UBS Securities Australia Ltd Buy 6.099 7.24 UBS Securities Australia Ltd Buy 6.099 7.24 UBS Securities Australia Ltd Buy 6.090 7.24 UBS Securities Australia Ltd Buy 6.090 7.24 UBS Securities Australia Ltd Buy </td <td>26-Mar-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>555</td> <td>29</td> <td>Ordinary</td>	26-Mar-14	UBS Securities Australia Ltd	Buy	555	29	Ordinary
UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 21,323 2,569 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 26,734 3,271 UBS Securities Australia Ltd Buy 26,734 3,73 UBS Securities Australia Ltd Buy 5,595 795 UBS Securities Australia Ltd Buy 5,609 3,736 UBS Securities Australia Ltd Buy 6,509 7,24 UBS Securities Australia Ltd Buy 6,509 7,24 UBS Securities Australia Ltd Buy 6,509 7,24 UBS Securities Australia Ltd Buy 6,509 7,37 UBS Securities Australia Ltd Buy 47,401 5,711 UBS Securities Australia Ltd Buy 6,009 7,32 UBS Securities Australia Ltd Buy 6,009 7,32 UBS Securities Australia Ltd Buy	26-Mar-14	UBS Securities Australia Ltd	Buy	638	77	Ordinary
UBS Securities Australia Ltd Buy 144 USB Securities Australia Ltd Buy 21,323 2,569 USB Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 26,734 370 UBS Securities Australia Ltd Buy 5,695 795 UBS Securities Australia Ltd Buy 5,695 795 UBS Securities Australia Ltd Buy 19,820 2,388 UBS Securities Australia Ltd Buy 6,595 754 UBS Securities Australia Ltd Buy 6,090 3,736 UBS Securities Australia Ltd Buy 6,090 3,736 UBS Securities Australia Ltd Buy 6,090 7,24 UBS Securities Australia Ltd Buy 6,090 7,34 UBS Securities Australia Ltd Buy 6,000 7,34 UBS Securities Australia Ltd Buy 6,000 7,35 UBS Securities Australia Ltd Buy 6,000 7,35 UBS Securities Australia Ltd Buy 6,000	26-Mar-14	UBS Securities Australia Ltd	Buy	600'9	724	Ordinary
UBS Securities Australia Ltd Buy 21,323 2,569 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 3,071 370 UBS Securities Australia Ltd Buy 26,734 3,221 UBS Securities Australia Ltd Buy 5,64 68 UBS Securities Australia Ltd Buy 31,009 3,756 UBS Securities Australia Ltd Buy 5,609 7,54 UBS Securities Australia Ltd Buy 4,740 5,711 UBS Securities Australia Ltd Buy 6,009 7,24 UBS Securities Australia Ltd Buy 4,740 5,711 UBS Securities Australia Ltd Buy 6,009 7,24 UBS Securities Australia Ltd Buy 3,629 6,377 UBS Securities Australia Ltd Buy 5,714 5,377 UBS Securities Australia Ltd Buy 4,32 2,59 UBS Securities Australia Ltd Buy 4,441 5,35 UBS Securities Australia Ltd Buy	26-Mar-14	UBS Securities Australia Ltd	Buv	946	114	Ordinary
UBS Securities Australia Ltd Buy 6,399 795 UBS Securities Australia Ltd Buy 5,071 370 UBS Securities Australia Ltd Buy 26,734 3,071 UBS Securities Australia Ltd Buy 5,69 785 UBS Securities Australia Ltd Buy 5,69 785 UBS Securities Australia Ltd Buy 31,009 3,736 UBS Securities Australia Ltd Buy 6,599 7,74 UBS Securities Australia Ltd Buy 6,009 7,74 UBS Securities Australia Ltd Buy 6,009 7,74 UBS Securities Australia Ltd Buy 6,009 7,37 UBS Securities Australia Ltd Buy 6,1063 7,357 UBS Securities Australia Ltd Buy<	26-Mar-14	UBS Securities Australia Ltd	Bily	21 373	2 5,60	Ordinan
UBS Securities Australia Ltd Buy 50,734 370 UBS Securities Australia Ltd Buy 56,734 3,221 UBS Securities Australia Ltd Buy 6,595 795 UBS Securities Australia Ltd Buy 6,595 795 UBS Securities Australia Ltd Buy 1,009 3,736 UBS Securities Australia Ltd Buy 6,595 795 UBS Securities Australia Ltd Buy 6,595 771 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 5,711 440 UBS Securities Australia Ltd Buy 5,292 6,377 UBS Securities Australia Ltd Buy 5,292 6,377 UBS Securities Australia Ltd Buy 7,52 995 UBS Securities Australia Ltd Buy 7,74 105 UBS Securities Australia Ltd Buy 6,404 7,890 UBS Securities Australia Ltd Buy 6,404 7,76 UBS Securities Australia Ltd Buy	26-Mar-14	(BC Countition Australia 144	6	0010	100,	Coldinary
UBS Securities Australia Ltd Buy 26,734 3.70 UBS Securities Australia Ltd Buy 26,734 3.20 UBS Securities Australia Ltd Buy 564 68 UBS Securities Australia Ltd Buy 564 68 UBS Securities Australia Ltd Buy 56,595 736 UBS Securities Australia Ltd Buy 6,595 736 UBS Securities Australia Ltd Buy 6,595 724 UBS Securities Australia Ltd Buy 6,509 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,106 7,357 UBS Securities Australia Ltd Buy 6,106 7,357 UBS Securities Australia Ltd Buy 8,259 95 UBS Securities Australia Ltd Buy 4,441 5,357 UBS Securities Australia Ltd Buy 6,500 7,357 UBS Securities Australia Ltd Buy 6,500 7,75 UBS Securities Australia Ltd Buy <td< td=""><td>26 kdor 1.4</td><td>The Countries Australia Ltd</td><td>, and</td><td>55000</td><td>CE/</td><td>Orginary</td></td<>	26 kdor 1.4	The Countries Australia Ltd	, and	55000	CE/	Orginary
UBS Securities Australia Ltd Buy 26,734 3,221 UBS Securities Australia Ltd Buy 3,071 395 UBS Securities Australia Ltd Buy 5,695 756 UBS Securities Australia Ltd Buy 3,000 3,786 UBS Securities Australia Ltd Buy 6,595 724 UBS Securities Australia Ltd Buy 6,009 7,24 UBS Securities Australia Ltd Buy 6,009 7,24 UBS Securities Australia Ltd Buy 6,009 7,24 UBS Securities Australia Ltd Buy 8,29,20 6,377 UBS Securities Australia Ltd Buy 8,29,20 6,377 UBS Securities Australia Ltd Buy 8,29,20 6,377 UBS Securities Australia Ltd Buy 8,259 6,39 UBS Securities Australia Ltd Buy 4,441 5,55 UBS Securities Australia Ltd Buy 6,54 7,73 UBS Securities Australia Ltd Buy 6,54 7,85 UBS Securities Australia Ltd Buy </td <td>- [</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>3,071</td> <td>370</td> <td>Ordinary</td>	- [UBS Securities Australia Ltd	Buy	3,071	370	Ordinary
UBS Securities Australia Ltd Buy 3,071 370 UBS Securities Australia Ltd Buy 6,595 795 UBS Securities Australia Ltd Buy 5,64 68 UBS Securities Australia Ltd Buy 6,595 736 UBS Securities Australia Ltd Buy 6,595 738 UBS Securities Australia Ltd Buy 6,509 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 5,711 7357 UBS Securities Australia Ltd Buy 6,009 7,357 UBS Securities Australia Ltd Buy 8,259 95 UBS Securities Australia Ltd Buy 4,441 5,55 UBS Securities Australia Ltd Buy 4		UBS Securities Australia Ltd	Buy	26,734	3,221	Ordinary
UBS Securities Australia Ltd Buy 6.595 795 UBS Securities Australia Ltd Buy 31,009 37.36 UBS Securities Australia Ltd Buy 19,820 2,388 UBS Securities Australia Ltd Buy 47,401 571 UBS Securities Australia Ltd Buy 6,099 724 UBS Securities Australia Ltd Buy 6,090 724 UBS Securities Australia Ltd Buy 6,090 7,357 UBS Securities Australia Ltd Buy 6,1,063 7,357 UBS Securities Australia Ltd Buy 6,1,063 7,357 UBS Securities Australia Ltd Buy 6,296 7,95 UBS Securities Australia Ltd Buy 6,5404 7,80 UBS Securities Australia Ltd <td< td=""><td></td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,071</td><td>370</td><td>Ordinary</td></td<>		UBS Securities Australia Ltd	Buy	3,071	370	Ordinary
UBS Securities Australia Ltd Buy 564 68 UBS Securities Australia Ltd Buy 31,009 3,736 UBS Securities Australia Ltd Buy 6,595 734 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 5,711 UBS Securities Australia Ltd Buy 5,222 6,377 UBS Securities Australia Ltd Buy 5,292 6,377 UBS Securities Australia Ltd Buy 1,632 5,27 UBS Securities Australia Ltd Buy 1,735 209 UBS Securities Australia Ltd Buy 4,32 52 UBS Securities Australia Ltd Buy 7,75 95 UBS Securities Australia Ltd Buy 7,75 95 UBS Securities Australia Ltd Buy 6,404 7,890 UBS Securities Australia Ltd Buy 4,411 5,357 UBS Securities Australia Ltd Buy 6,404 7,890 UBS Securities Australia Ltd Buy 6,304 <td< td=""><td></td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>6.595</td><td>795</td><td>Ordinary</td></td<>		UBS Securities Australia Ltd	Buy	6.595	795	Ordinary
UBS Securities Australia Ltd Buy 31,009 3,736 UBS Securities Australia Ltd Buy 1,9820 2,388 UBS Securities Australia Ltd Buy 6,599 795 UBS Securities Australia Ltd Buy 47,401 5,711 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 6,009 737 UBS Securities Australia Ltd Buy 6,063 7,357 UBS Securities Australia Ltd Buy 6,063 7,357 UBS Securities Australia Ltd Buy 8,259 995 UBS Securities Australia Ltd Buy 8,259 995 UBS Securities Australia Ltd Buy 6,504 7,880 UBS Securities Australia Ltd Buy 6,5404 7,880 UBS Securities Australia Ltd	ŀ	UBS Securities Australia I to	Bir	252	a	Ordingo
UBS Securities Australia Ltd Buy 19,802 2,388 UBS Securities Australia Ltd Buy 6,595 795 UBS Securities Australia Ltd Buy 6,595 771 UBS Securities Australia Ltd Buy 6,000 7,24 UBS Securities Australia Ltd Buy 6,000 7,24 UBS Securities Australia Ltd Buy 6,1063 7,357 UBS Securities Australia Ltd Buy 4,32 5,2 UBS Securities Australia Ltd Buy 4,32 5,2 UBS Securities Australia Ltd Buy 8,259 995 UBS Securities Australia Ltd Buy 7,735 109 UBS Securities Australia Ltd Buy 7,735 995 UBS Securities Australia Ltd Buy 6,404 7,880 UBS Securities Australia Ltd Buy 6,404 7,880 UBS Securities Australia Ltd Buy 6,377 3,600 UBS Securities Australia Ltd Buy 6,377 3,600 UBS Securities Australia Ltd Buy	Ł	Hac Countition Australia 144		000 10	200	o deliai
UBS Securities Australia Ltd Buy 6 598 7.98 UBS Securities Australia Ltd Buy 6 599 724 UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 5,711 47,401 5,711 UBS Securities Australia Ltd Buy 5,2,329 6,377 405 UBS Securities Australia Ltd Buy 51,623 7,357 52 UBS Securities Australia Ltd Buy 8,259 95 52 UBS Securities Australia Ltd Buy 8,259 95 95 95 UBS Securities Australia Ltd Buy 8,259 95	ti i i i i i i	Opp Securities Australia Ltd	buy	600,18	3,/30	Ordinary
UBS Securities Australia Ltd Buy 6,595 795 UBS Securities Australia Ltd Buy 4,040 5,711 UBS Securities Australia Ltd Buy 6,009 7,724 UBS Securities Australia Ltd Buy 5,952 440 UBS Securities Australia Ltd Buy 5,052 6,377 UBS Securities Australia Ltd Buy 61,063 7,357 UBS Securities Australia Ltd Buy 4,32 5,09 UBS Securities Australia Ltd Buy 4,441 5,50 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 63,704 7,880 UBS Securities Australia Ltd Buy 63,704 7,880 UBS Securities Australia Ltd Buy 63,704 7,880 UBS Securities Australia Ltd	26-Mar-14	UBS securities Australia Ltd	Buy	19,820	2,388	Ordinary
UBS Securities Australia Ltd Buy 47,401 5,711 UBS Securities Australia Ltd Buy 5,609 724 UBS Securities Australia Ltd Buy 5,529 6,377 UBS Securities Australia Ltd Buy 1,633 2,929 6,377 UBS Securities Australia Ltd Buy 1,537 209 UBS Securities Australia Ltd Buy 4,32 52 UBS Securities Australia Ltd Buy 7,55 95 UBS Securities Australia Ltd Buy 7,75 9 UBS Securities Australia Ltd Buy 65,404 7,890 UBS Securities Australia Ltd Buy 30,627 3,690 UBS Securities Australia Ltd Sell 1,890 (1,807) UBS Securities Australia Ltd Sell 2,704 (1,807) UBS Securities Australia	26-Mar-14	UBS Securities Australia Ltd	Buy	6,595	795	Ordinary
UBS Securities Australia Ltd Buy 6,009 724 UBS Securities Australia Ltd Buy 3,652 440 UBS Securities Australia Ltd Buy 6,1,063 7,357 UBS Securities Australia Ltd Buy 6,1,063 7,357 UBS Securities Australia Ltd Buy 4,259 995 UBS Securities Australia Ltd Buy 8,259 995 UBS Securities Australia Ltd Buy 4,441 5,80 UBS Securities Australia Ltd Buy 65,404 7,80 UBS Securities Australia Ltd Buy 63,1 76 UBS Securities Australia Ltd Sell 2,704 (3,007) UBS Securities Australia Ltd Sell 1,500 (1,007) UBS Securities Australia Ltd Sell (1,007) (1,007)	26-Mar-14	UBS Securities Australia Ltd	Buy	47,401	5,711	Ordinary
UBS Securities Australia Ltd Buy 3,652 440 UBS Securities Australia Ltd Buy 5,329 6,377 UBS Securities Australia Ltd Buy 61,063 7,357 UBS Securities Australia Ltd Buy 8,259 995 UBS Securities Australia Ltd Buy 7,75 995 UBS Securities Australia Ltd Buy 7,75 995 UBS Securities Australia Ltd Buy 6,404 7880 UBS Securities Australia Ltd Buy 6,404 7880 UBS Securities Australia Ltd Buy 6,504 7,880 UBS Securities Australia Ltd Buy 6,504 7,880 UBS Securities Australia Ltd Buy 6,537 3,690 UBS Securities Australia Ltd Sell 1,830 (180) UBS Securities Australia Ltd Sell 8,704 (180) UBS Securities Australia Ltd Sell 1,500 (180)	26-Mar-14	UBS Securities Australia Ltd	Buy	600'9	724	Ordinary
UBS Securities Australia Ltd Buy 52,929 6,377 UBS Securities Australia Ltd Buy 61,063 7,357 UBS Securities Australia Ltd Buy 432 52 UBS Securities Australia Ltd Buy 8,259 955 UBS Securities Australia Ltd Buy 8,259 955 UBS Securities Australia Ltd Buy 4,441 5,35 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 6,50 7,70 UBS Securities Australia Ltd Sell 1,83 (22) UBS Securities Australia Ltd Sell 1,83 (180) UBS Securities Australia Ltd Sell 1,500 (180)	26-Mar-14	UBS Securities Australia Ltd	Buv	3.652	440	Ordinary
UBS Securities Australia Ltd Buy 61,063 7,357 UBS Securities Australia Ltd Buy 1,735 2.09 UBS Securities Australia Ltd Buy 8,259 995 UBS Securities Australia Ltd Buy 75 9 UBS Securities Australia Ltd Buy 4,41 535 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Sell 1,500 (180) UBS Securities Australia Ltd Sell 8,376 (180) UBS Securities Australia Ltd Sell 1,500 (180)	26-Mar-14	URS Sequitities Australia 11d	Birk	52 020	7753	Ordinary
UBS Securities Australia Ltd Buy 1,783 7,357 UBS Securities Australia Ltd Buy 432 52 UBS Securities Australia Ltd Buy 8,259 95 UBS Securities Australia Ltd Buy 7,5 95 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 631 76 UBS Securities Australia Ltd Buy 631 76 UBS Securities Australia Ltd Sell 183 (22) UBS Securities Australia Ltd Sell 1,704 (180) UBS Securities Australia Ltd Sell 2,704 (32) UBS Securities Australia Ltd Sell (1,007) UBS Securities Australia Ltd Sell (1,800)	26 Mar 14	1 IDC Countries Australia 14d	626	61,925	1,2,0	Sign of the second
UBS Securifies Australia Ltd Buy 4.735 209 UBS Securifies Australia Ltd Buy 4.32 5.25 UBS Securifies Australia Ltd Buy 8,259 995 UBS Securifies Australia Ltd Buy 4,441 5.35 UBS Securifies Australia Ltd Buy 65,404 7,880 UBS Securifies Australia Ltd Sell 1,704 (3,27) UBS Securifies Australia Ltd Sell 2,704 (3,25) UBS Securifies Australia Ltd Sell 1,500 (1,800) UBS Securifies Australia Ltd Sell 1,500 (1,800)	T L L L L L L L L L L L L L L L L L L L	ODS Securities Australia Ltd	any	61,063	/55'/	Ordinary
UBS Securities Australia Ltd Buy 432 52 UBS Securities Australia Ltd Buy 8,259 995 UBS Securities Australia Ltd Buy 75 9 UBS Securities Australia Ltd Buy 4,41 535 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,70 3,690 UBS Securities Australia Ltd Sell 1,704 (180) UBS Securities Australia Ltd Sell 8,704 (180) UBS Securities Australia Ltd Sell 8,704 (180)	26-Mar-14	UBS Securities Australia Ltd	Buy	1,735	509	Ordinary
UBS Securities Australia Ltd Buy 8,259 995 UBS Securities Australia Ltd Buy 775 9 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 63,1 76 UBS Securities Australia Ltd Sell 1,83 (22) UBS Securities Australia Ltd Sell 2,704 (325) UBS Securities Australia Ltd Sell 2,704 (325) UBS Securities Australia Ltd Sell (1,807) UBS Securities Australia Ltd Sell (1,800)	26-Mar-14	UBS Securities Australia Ltd	Buy	432	52	Ordinary
UBS Securities Australia Ltd Buy 75 9 UBS Securities Australia Ltd Buy 4,441 535 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 63,71 76 UBS Securities Australia Ltd Sell 3,690 183 UBS Securities Australia Ltd Sell 2,704 (32) UBS Securities Australia Ltd Sell 2,704 (32) UBS Securities Australia Ltd Sell 2,704 (32) UBS Securities Australia Ltd Sell 2,704 (180) UBS Securities Australia Ltd Sell 1,500 (180)	26-Mar-14	UBS Securities Australia Ltd	Buy	8,259	995	Ordinary
UBS Securities Australia Ltd Buy 4,441 535 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,27 3,690 UBS Securities Australia Ltd Sell 183 (22) UBS Securities Australia Ltd Sell 2,704 (32) UBS Securities Australia Ltd Sell 8,378 (180) UBS Securities Australia Ltd Sell 1,500 (180)	26-Mar-14	UBS Securities Australia Ltd	Buy	75	6	Ordinary
UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 65,404 7,880 UBS Securities Australia Ltd Buy 30,627 3,690 UBS Securities Australia Ltd Sell 1,83 (22) UBS Securities Australia Ltd Sell 2,704 (325) UBS Securities Australia Ltd Sell 8,378 (1,007) UBS Securities Australia Ltd Sell 1,500 (1,800)	26-Mar-14	UBS Securities Australia Ltd	Buv	4.441	535	Ordinary
UBS Securities Australia Ltd Buy 631 76 UBS Securities Australia Ltd Buy 30,627 3,690 UBS Securities Australia Ltd Sell 2,704 (22) UBS Securities Australia Ltd Sell 2,704 (35) UBS Securities Australia Ltd Sell 3,704 (35) UBS Securities Australia Ltd Sell 1,500 (180)	26-Mar-14	UBS Securities Australia Ltd	Buy	65,404	7 880	Ordinan
UBS Securities Australia Ltd Buy 30,627 3,690 UBS Securities Australia Ltd Sell 183 (22) UBS Securities Australia Ltd Sell 2,704 (32) UBS Securities Australia Ltd Sell 8,378 (1,007) UBS Securities Australia Ltd Sell 1,500 (180)	26-Mar-14	UBS Securities Australia I to	Rusy	169	27	Ordinary
UBS Securities Australia Ltd Sell 5,590 UBS Securities Australia Ltd Sell 2,704 UBS Securities Australia Ltd Sell 2,704 UBS Securities Australia Ltd Seli 8,378 UBS Securities Australia Ltd Seli 1,500	25 Mar 14	THE STREET WASHINGTON	, and	150	0/	Cidinary
UBS Securities Australia Ltd Sell 183 (22) UBS Securities Australia Ltd Sell 2,704 (355) UBS Securities Australia Ltd Sell 2,704 (355) UBS Securities Australia Ltd Sell (1,500) (180)	50-INIGIT 14	Obs securities Australia Ltd	Buy	30,527	3,590	Ordinary
UBS Securities Australia Ltd Seli 2,704 (325) UBS Securities Australia Ltd Seli 8,378 (1,007) UBS Securities Australia Ltd Seli 1,500 (180)	26-Mar-14	UBS Securities Australia Ltd	Sell	183	(22)	Ordinary
UBS Securities Australia Ltd Seli 8,378 (1,007) UBS Securities Australia Ltd Sell 1,500 (180)	26-Mar-14	UBS Securities Australia Ltd	Seli	2,704	(322)	Ordinary
UBS Securities Australia Ltd Sell (180)	26-Mar-14	UBS Securities Australia Ltd	Seli	8.378	(1.007)	Ordinary
COO COCCULIAGES AGENT FILE	26-Mar-14	I IRS Societies Asistralia I to	=			
	* 1 - 1 A			- 500	(100)	, independ

UBS Securities Australia Ltd	Į,			
UBS Securities Australia Ltd	Sell	612	(72)	Ordinary
UBS Securities Australia Ltd	Sell	1,311	(154)	Ordinary
S Securities Australia Ltd	Seli	3,476	(408)	Ordinary
JBS Securities Australia Ltd	Sell	1,447	(170)	Ordinary
JBS Securities Australia Ltd	Sell	1,702	(200)	Ordinary
UBS Securities Australia Ltd	Sell	162	(61)	Ordinary
UBS Securities Australia Ltd	Sell	179	(21)	Ordinary
UBS Securities Australia Ltd	Sell	689	(81)	Ordinary
S Securities Australia Ltd	Sell	1.532	(180)	Ordinary
UBS Securities Australia Ltd	Seli	868 5	(457)	Ordinary
S Securities Australia 11d	Coll	100	(52.4)	Cadipan
Contraction Australia (+a)	1100	555,1	(111)	Cignial y
COO Seculates Australia Etc	Jell C	714,1	(001)	Orginary
S Securities Australia LtG	Sell	798,1	(6LZ)	Ordinary
JBS Securities Australia Ltd	Sell	1,429	(168)	Ordinary
UBS Securities Australia Ltd	Sell	1,037	(122)	Ordinary
S Securities Australia Ltd	Sell	4.533	(533)	Ordinary
UBS Securities Australia (1)	la)	3 120	(367)	Ordinan
IBS Contrition Australia Ltd	100	4 177	(100)	
Secondary August Line	100	4,177	(452)	Comary
Jobs Securities Australia Ltd	Neil Neil	3,112	(36/)	Ordinary
S Securities Australia Ltd	Sell	4,359	(514)	Ordinary
S AG Australia Branch	Sell	240,838	(28,754)	Ordinary
UBS AG Australia Branch	Riv	220.458	26 422	Ordinary
The state of the s	50	9C#1022	774'07	Oldinary
JBS AG Australia Branch	Buy	2,287,543	273,985	Ordinary
JBS AG Australia Branch	Buy	310,834	37,229	Ordinary
UBS AG Australia Branch	Buy	79 954	3 590	Ordinary
BS AG Australia Branch	Bir	908 09	2778	Ordingo
O A G Australia Branch		500,200	2000	diam'r.
DOS ACI Acidada piatrol	and .	(5,7,2,1	808,08	Ordinary
S AG London Branch	Sell	1,156,458	(138,134)	Ordinary
JBS AG London Branch	Buy	1,162,706	138,134	Ordinary
UBS Securities Australia Ltd	Buy	32.153	3.811	Ordinary
IBS Securities Australia Itd	9	67 147	(USU a)	Ordinan
The state of the s	120	(4)	(000'0)	Oldinary
UBS Securities Australia Ltd	Buy	24,930	3,000	Ordinary
UBS Securities Australia Ltd	Buy	1,974,584	236,548	Ordinary
S Securities Australia Ltd	Sell	1,665,025	(199,593)	Ordinary
S Securities Australia Ltd	Buv	40.708	4.881	Ordinary
JBS Securities Australia Ltd	lle?	707 59	(7.5.77)	Ordinan
DO Contration Australia 144		200	7,000	O Colonia
Decalities Australia Liu	for	71,14	24,093	Ordinary
S Securities Australia Ltd	Sell	18,538	(2,208)	Ordinary
UBS Securities Australia Ltd	Sell	1,156,458	(138,134)	Ordinary
UBS Securities Australia Ltd	Sett	3.573	(431)	Ordinary
JBS Securities Australia Ltd	les	3 163	(282)	Ordinary
		501,0	(205)	o domain
UBS Securities Australia Ltd	sell	3,681	(447)	Ordinary
JBS Securities Australia Ltd	Sell	3,234	(393)	Ordinary
S Securities Australia Ltd	Sell	1.572	(192)	Ordinary
UBS Securities Australia Ltd	Sel	4676	(1271)	Ordinan
CO Contraction August 1 and	1-0	2007	(0.00)	0.00
Obs securities Australia Ltd	Sell	3,104	(3/8)	Ordinary
JBS Securities Australia Ltd	Sell	4,272	(521)	Ordinary
UBS Securities Australia Ltd	Sell	3,268	(368)	Ordinary
UBS Securities Australia Ltd	lay	100	(121)	Ordinary
IBS Congrition Australia (4d	3		(4)	
Securities Australia Ltd	Dell.	80	=	Ordinary
UBS Securities Australia Ltd	Se	9,156	(1,122)	Ordinary
UBS Securities Australia Ltd	Buy	908'999	80,435	Ordinary
UBS Securities Australia Ltd	3	78 084	(9 419)	Ordinary
10C Contration Augustin 14d	1 0	000	(2004)	
o securities Australia Liu	II DC	3,000	(060/1)	Ordinary
UBS Securities Australia Ltd	Buy	7,212	820	Ordinary
S Securities Australia Ltd	ll as	234,963	(28,343)	Ordinary
JBS Securities Australia Ltd	lles	090 9	(731)	Ordinary
IRS Sequirities Australia Ltd	Coll	46 147	(5 4 46)	Ordinan
DC Commission Assessed and	1-3	151,55	(00000)	i i
S Securities Australia Ltd	Sell	65,781	(466'/)	Ordinary
UBS Securities Australia Ltd	Sell	14,209	(1,714)	Ordinary
JBS Securities Australia Ltd	Sell	1691	(204)	Ordinary
IDO Constitute Australia 144	30	140	(101)	Tall Co
Securities Australia Ltd	Sell	149	(8)	Orginary
JBS Securities Australia Ltd	Sell	32,339	(3,901)	Ordinary
UBS Securities Australia Ltd	Buy	385,170	46,462	Ordinary
LIBC Constitute Assetsolis 14d	Coll			
o seculities Australia Liu	1		14-4/	
		4/2	(23)	Ordinary

111-11					
26 May 14	UBS Securities Australia Ltd	Buy	250	30	Ordinary
p-ividi-14	Ubs securities Australia Ltd	Buy	208	19	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	1,015	122	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buy	2,022	243	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	4,052	487	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buv	8.095	973	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Briv	16 199	1 047	Ordinan
26-Mar-14	LIBS Securities Australia Ltd	Bio	25,037	000	Croling
6-Mar-14	I IBS Securities Australia 14d	, and	0300	20,0	9
Mar 14	I IRC Counties Australia 1+4	900	2000,5	001,1	Odinary
- Par - 22	COO Securities Australia Liu	Buy	/05/66	4,323	Orginary
-Mar-14	UBS Securities Australia Ltd	Buy	32,398	3,894	Ordinary
b-Mar-14	UBS Securities Australia Ltd	Buy	64,247	7,722	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	9,385	1,128	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	541	9	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buv	47.840	5.750	Ordinary
-Mar-14	UBS Securities Australia 11d	Biv	65,728	7 900	Ordingo
26-Mar-14	I BS Securities Australia I tol	67 4	22,000	2006,	Cidinaly
Mar 14	I I DC Coo mitton A contact to 1 to 1	ána a	604,12	400,0	Ordinary
t law o	Opp permittes Australia Liu	Buy	OC.	٥	Ordinary
th JPIN	Upo Securities Australia Ltd	Buy	967'7	9/7	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buy	3,927	472	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	40,577	4,877	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	62,608	7,525	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	12,871	1.547	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buy	7,962	256	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buv	18.096	2175	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	67	α	Ordinan
-Mar-14	UBS Securities Australia I td	Biv	39 104	A 700	Ording
6-Mar-14	I IBS Sequition Australia 14d	G G	T21,55	2,,,	ordinal y
	COO Securities Australia Lite	fog :	/ / / / /	2,133	Orginary
20-Mar-14	UBS Securities Australia Ltd	Buy	21,349	2,566	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buy	120,390	14,470	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	2,238	569	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buy	50,436	6.062	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	115,398	13.870	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buv	17.996	2 163	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buv	10.808	1 299	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Bis	26.832	3 2 2 5	Coding
26-Mar. 14	I IBS Sequitities Australia 14d	Billi	2,147	0110	in in
1000	The Country Against Life	, and	7,1147	730	Clumary
Mar-14	UBS Securities Australia Ltd	Buy	3,120	375	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buy	42,698	5,132	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	2,496	300	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Biv	7197	SEE	Ordinary
6 Mar 14	Libo Counting Asserted	650	1000	3	y digital y
Mar-14	Ups securities Australia Ltd	Buy	/,604	914	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	29	00	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buy	150,243	18,058	Ordinary
-Mar-14	UBS Securities Australia Ltd	Buy	41.434	4.980	Ordinary
26-Mar-14	UBS Securities Australia I to	Bliv	3 253	100	Ordinan
26-Mar-14	I IBS Securities Australia 14d	Price	929 01	000	or in any
26 May 14	I IDC Convision August a 1-1	652	200,04	500,4	S C C C C C C C C C C C C C C C C C C C
t lalal	ODS Securities Australia Liu	buy	706	20.0	Ordinary
o-Mar-14	UBS Securities Australia Ltd	Buy	20,525	2,467	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	37,565	4,515	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	24,111	2,898	Ordinary
26-Mar-14	UBS Securities Australia Ltd	N/N/S	34.445	4 140	Ordinary
26-Mar-14	II IRS Sequities Australia #4	8187	50 515	361.0	, acaipo
		000	Cicion	0,433	Cidinaly
40-IVIAI - 14	ODS Seculifies Australia Ltd	eny	090,52	3,012	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	22,431	2,696	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	26,250	3,155	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Buy	25,634	3,081	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	63.989	7.691	Ordinary
26-Mar-14	ILIRS Securities Australia Ltd	, id	65 370	7.957	Ording
11.11	Little Contract Acceptance	(5.4	750 55	(50,7	6 5 6
ZO-INIGIT-14	Obs securities Australia Ltd	pay	0/6///	9,372	Ordinary
Mar-14	UBS Securities Australia Ltd	Buy	29,461	3,541	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	35,418	4,257	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	24,968	3.001	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Bit	11 117	, , ,	
			777	4 481	Ordinary
V + + 4	Late Carried and Control of the Cont	503	41,444	4,981	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Buy	36,533	4,391	Ordinary

UBS Securities Australia Ltd	Buy	10,621	7/7	Clamary
UBS Securities Australia Ltd	Buy	30,678	3,674	Ordinary
UBS Securities Australia Ltd	Buy	3,090	370	Ordinary
UBS Securities Australia Ltd	Buy	4,618	223	Ordinary
a Ltd	Buy	16,700	2,000	Ordinary
UBS Securities Australia Ltd	Buy	33,692	4,035	Ordinary
UBS Securities Australia Ltd	Buy	18,737	2,244	Ordinary
UBS Securities Australia Ltd	Buy	21,794	2,610	Ordinary
UBS Securities Australia Ltd	Buy	31,287	3,747	Ordinary
a Ltd	Buy	24,549	2,940	Ordinary
UBS Securities Australia Ltd	Buy	19,230	2,303	Ordinary
UBS Securities Australia Ltd	Buy	3,507	420	Ordinary
UBS Securities Australia Ltd	Buy	18,779	2,249	Ordinary
UBS Securities Australia Ltd	Buy	42.493	5.089	Ordinary
JBS Securities Australia Ltd	Buv	11 966	1 433	Ordinary
185 Securities Australia I to	19	6.404	(7,67)	, decipio
IBS Securities Australia 14d	Col	A 001	(503)	Ordina
		355,5	(150)	O'CILIBIA
2	100	0/7/0	(167)	Ordinary
UBS Securities Australia Ltd	<u> </u>	3,586	(430)	Ordinary
UBS Securities Australia Ltd	Sell	5,085	(609)	Ordinary
ס	<u></u>	4,528	(245)	Ordinary
P	Sell	4,050	(485)	Ordinary
JBS Securities Australia Ltd	J	2 961	(355)	Ordinary
IBC Sequition Australia Ltd	Col	010	(103)	Ordinan
DOC Commission Assessment and	27.0	500 0	(21)	i i i
	Sei	7,037	(244)	Ordinary
Securities Australia Ltd	Sell	1,645	(197)	Ordinary
JBS Securities Australia Ltd	Sell	3,717	(446)	Ordinary
JBS Securities Australia Ltd	Sell	3,240	(68E)	Ordinary
JBS Securities Australia Ltd	Sell	2.296	(275)	Ordinary
JBS Securities Australia 11d	الع	2 143	(757)	Ordinary
IBS Son prities Australia 14d	Sell	1 905	(717)	Ordinary
	110	305.5	100	0
	77	202 6	140	i i i i i
DC Commission American Lead		000/0	55.	or or or
DDS Securities Australia Ltd	, and	11,324	198,1	Ordinary
DDS Securities Australia Ltu	Áng	7/5'1	681	Grainary
	Buy	109,408	13,150	Ordinary
JBS Securities Australia Ltd	Buy	59,230	7,119	Ordinary
UBS Securities Australia Ltd	Buy	19,011	2,285	Ordinary
þ	Buy	64,322	7,731	Ordinary
70	Buy	36,333	4,367	Ordinary
þ	Bc	24.136	2.901	Ordinary
LIBS Securities Australia I td	Birv	36 908	4436	Ordinary
IDC Cognition Australia 14d		00,00	000,4	O Grandly
2 .	, and	671,00	070'/	Ordinary
Obs securities Australia Ltd	Buy	24,585	cc6'7	Ordinary
to.	Buy	36,783	4,421	Ordinary
UBS Securities Australia Ltd	Buy	35,917	4,317	Ordinary
2	Buy	36,275	4,360	Ordinary
UBS Securities Australia Ltd	Buy	6.393	1.129	Ordinary
-	Biz	22,689	2 777	Ordinary
DC Contraction Australia 144	, ,	36 467	COC V	
	600	,0±'00	100,4	Cidinaly
ODS Securifies Australia Ltd	ans	77,614	2'/18	Ordinary
UBS Securities Australia Ltd	Buy	35,327	4,246	Ordinary
_	Buy	65,953	7,927	Ordinary
UBS Securities Australia Ltd	Buy	20.700	2.488	Ordinary
	N. S.	27 389	3 202	Ordinan
100 Consisting Assessed 14d	20	25,727	2000	
	, and	01/0/	2,220	Clulialy
UBS Securities Australia Ltd	Buy	33,521	4,029	Ordinary
	Buy	11,981	1,440	Ordinary
JBS Securities Australia Ltd	Buy	17,314	2,081	Ordinary
JBS Securities Australia Ltd	Buy	43,747	5,258	Ordinary
UBS Securities Australia Ltd	Buy	37.931	3 958	Ordinary
IBS Sequitible Australia Ltd	N. W.	1 373	150	Ordinan
	Ben	57.5 CE	2 870	Ordinary
IDO Consister Australia I+A		10,000	2000	Oldinary Ordinary
COO COMMENT A CALL TO THE	57	020,61	7,200	Ordinary
	-			
		906,1	(525)	Ordinary

A 1 4 2 4 4	HIDE Commission Assessed and			1	
6-Mar-14	UBS Securities Australia Itd	= X	0077 55	4 0000	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Sell	2,617	(313)	Ordinal
6-Mar-14	UBS Securities Australia Ltd	Sell	612	(2.5)	Ordinary
26-Mar-14	UBS Securities Australia I to	9	202	(0)()	Ordinary
6-Mar. 14	I IRS Securities Australia 144	3	410,2	(000)	S C C C C C C C C C C C C C C C C C C C
26 Mar 14	Libe Counting Australia Lid	120	010	(76)	Orginary
- Initial - 12	COO Deculines Australia Eu	EX.	(,/45	(208)	Ordinary
#I-IIII	Obs Securities Australia Ltd	Sell	3,587	(426)	Ordinary
b-Mar-14	UBS Securities Australia Ltd	Sell	34	(4)	Ordinary
5-Mar-14	UBS Securities Australia Ltd	Seli	2,529	(300)	Ordinary
5-Mar-14	UBS Securities Australia Ltd	Sell	396	(47)	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Sell	160	(19)	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Sell	2,682	(318)	Ordinary
26-Mar-14	UBS Securities Australia 11d	Soll	BOE	(101)	Ordinary
6-Mar-14	I IRS Securities Australia 144	130	1000	(101)	Cicinaly
1 1 1 2	U.D.C. Commission A. Caralle Lead	100	2,304	(7/7)	Ordinary
iniai - I -	Opp Securities Abstralia Ltd	liac l	905	(/OL)	Ordinary
o-Mar-14	UBS Securities Australia Ltd	Jes.	3,316	(391)	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Sell	1,187	(140)	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Sell	1,819	(214)	Ordinary
6-Mar-14	UBS Securities Australia Ltd	Sell	2,049	(241)	Ordinary
26-Mar-14	UBS Securities Australia Ltd	Seli	1,454	(171)	Ordinary
5-Mar-14	UBS Securities Australia Ltd	Sell	3511	(414)	Ordinary
6-Mar-14	UBS Securities Australia I td	الع	3 689	(151)	Ordinan
26-Mar-14	ILIBS Sequirities Australia Ltd	Soll	4 530	(003)	Ordings.
6-Mar-14	I IRS Securities Australia I to	100	3 460	(307)	o di la
K-hAsr-14	I IBC Contraction Automatical Co.	= 20	00404	(140)	Claira,
1 V V V	2	0	4,043	(343)	Oronnary
1	CO AC ACATAL CO	Buy	100,00	7,349	Ordinary
-Ividr-14	UDS AG Australia Branch	Buly	/18,684	87,500	Ordinary
7-Mar-14	UBS AG Australia Branch	Buy	1,334,699	162,500	Ordinary
7-Mar-14	UBS AG Australia Branch	Sell	206,386	(25,000)	Ordinary
'-Mar-14	UBS Securities Australia Ltd	Buy	22.700	2.700	Ordinary
'-Mar-14	UBS Securities Australia Ltd	Buv	5.425	299	Ordinary
7-Mar-14	I co	Buy	24 053	3 033	Ordinary
7-Mar-14	I ISC Societities Australia 140	6	000 100	104 150	Oldina)
ividi	Opp Securities Australia Ltd	Buy	080,188	104,568	Ordinary
/-INIdr-14	ODS Securities Australia Ltd	Sell	914,168	(108,757)	Ordinary
+I-IPINI-	Obs securities Australia Ltd	Sell	61,045	(1,381)	Ordinary
/-Mar-14	es.	Buy	20,400	2,500	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	34,374	(4,108)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Buy	114.928	13.735	Ordinary
7-Mar-14	LIBS Securities Australia Ltd	Bir	11 515	1 400	Ordinan
W. 12	LIDE Consister Australia Ltd	Section 1	616,11	004,	o della
# - IPIAI-/	Obsider Australia Ltd	Enter Into Derivative (Ϋ́	1,526	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	794,478	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (ΑN	75,820	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (Α/N	50 171	Ontions
-Mar-14	ILIRS Securities Australia Ltd	Enter Into Derivative	\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	101 041	cucion C
7 NASE 14	DO Contribute Assets 1-1	Curici alica Collinativa	(9)	10,10	Silono
thial - lat	Obs securities Australia Ltd	Errier into Derivative C	Y.	2,314,523	Options
/-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	ΑN	61,556	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (WA	75,820	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	4,465	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (Ϋ́Z	15,639	Options
'-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative O	Ϋ́	181.041	Options
7-Mar-14	II IRS Securities Australia (1)	Contention Designation	NIA.	074 470	
7 NATE 14	I IOC Constitution Assets In Lead	The last of the la	100	0/4/46/	options.
- I	CCO Securities Australia Ctu	Ellier IIIO Dell'Adime	2	076,1	Shortons
/-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	NA NA	112,941	Options
'-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	2,740	Options
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (ΝΑ	4,465	Options
7-Mar-14	UBS Securities Australia Ltd	Enter into Derivative (ΨZ.	112.941	Options
27-Mar-14	UBS Securities Australia Ltd	Foter into Derivative of	VN	2 314 673	sucitary.
7 140.5 14	1 IDC Constitute Australia 1 au	7	***	20,710,7	chondo
-INIGL -	Obsisted Hills Australia Lid	Enter Mio Derivative of	XX	1/1/05	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	NA	2,740	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	61,556	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	Ž	50.171	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative	Ψ/N	75,820	Ontions
Mar-14	LIBC Securities Australia 14d	Patent Into Contration of	VIIV	72,000	2 553
-Mar-14	UBS Securities Australia Ltd	I FOTO I DELIVATION CO	- 0000		
		S THE STATE OF THE	T/A	112,941	Options
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	N/A	(112,941	Options

Buy 29,004 Buy 41,542 Buy 41,542 Buy 41,542 Buy 12,005 Buy 12,005 Buy 12,005 Buy 12,005 Buy 40,5687 Buy 6,0287 Buy 6,119 Buy 6,130 Buy 1,523 Buy 1,523 Buy 1,523 Buy 1,523 Buy 1,523 Buy 1,523	3,486 Ordinary	L		L	855 Ordinary					557 Ordinary		366 Ordinary	L				021 Ordinary							031 Ordinary		271 Ordinary		736 Ordinary		469 Ordinary				212 Ordinary		339 Ordinary		183 Ordinary	╛	5 Ordinary						011 Ordinary		364 Ordinary	500 Ordinary	110 Ordinary	7 Ordinary			1		oss Ordinary	1	205 Ordinary	obz Ordinary		l	1	50 Ordinary		
																																										8 320											7						-						416 50 4,515 543
	Buy	Buý	Buy	Buy	Buy	Riny	, in a	euy	Buy	Buy	8uy	Buy	Buy	Buv	72.0	ng a	Buy	Buy	Виу	Buy	any	buy.	25	Buv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	sng.	Buy	buy Buy	A Series	Bit S	Sin A	D. 1.		Buy																			
그러리리리리리리리리리리리리리리리리리리리리리리리리리리리리티티티티티티티티티티	urities Australia Ltd	urities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	urities Australia I to	control Australia Ltd	urities Australia Ltd	unities Australia I tol	Thirds Australia City	urities Australia Ltd	uritles Australia Ltd	arties Australia Ltd	unities Australia Ltd	inties Australia Itd	urities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	urities Australia Ltd	urities Australia Ltd	JBS Securities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urties Australia Ltd	unties Australia Ltd	Jritles Australia Ltd	urties Australia Ltd	urities Australia Ltd	unties Australia Ltd	Julies Australia Liu	Jritles Australia Lto	rrities Australia Ltd	rities Australia Itd	urities Australia I td	infles Australia Itd	rities Australia I td	Allines Avenue and	UBS Securities Australia Ltd																											

4/-INIGI-14	UBS Securities Australia Ltd	Sell	3,129	(3/3)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	1,458	(174)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	292	(16)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	1,223	(146)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	3,578	(427)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	2,889	(345)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	2,509	(562)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	218	(56)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	2,047	(244)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	1.990	(737)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	1,828	(218)	Ordinary
27-Mar-14	I IRS Securities Australia I to	9	07.00	(1221)	Ordinan
27-Mar-14	UBS Securities Australia Ltd	9	2 15/1	(25.7)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	9	2 412	(050)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	3	3.050	(364)	Ordinary
7-Mar-14	I IBS Securities Australia I to	9	2,713	(505)	Ordinary
7-Mar-14	I IRS Securities Australia 11d	3	4 500	(570)	Ordinary
7-Mar-14	I IBS Secretives Australia (1)	100	1 424	(171)	Ordinary
7-Mar-14	I IRS Societies Australia 14d	lec.	1,434	(1/1)	Ordinary
7-Mar-14	I IRS Sequenties Australia Hd	Sell	3000	(302)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Self-	OGC O	(6)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	lle?	3 438	(017)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	1109	(192)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Se	3.557	(424)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	3,323	(397)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	1,096	(131)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	1,814	(217)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Setl	4,364	(522)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Seil	5,853	(701)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	4,556	(545)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	15,807	(1,893)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	20,382	(2,441)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	20,499	(2,455)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	14,179	(1,696)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Self	2,271	(272)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	11,152	(1,334)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	7,775	(026)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	12,749	(1,525)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	6,186	(740)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	7,432	(889)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	3,135	(375)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	11,737	(1,404)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	15,148	(1,812)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	026	(116)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	7,975	(954)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	965'9	(789)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	les	2,633	(315)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Seli	6,337	(758)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Seli	2,583	(309)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	11,771	(1,408)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	5,459	(653)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	6,337	(758)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	1,237	(148)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	2,274	(272)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	1,120	(134)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	7,194	(861)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Seil	5,718	(684)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	10,118	(1,211)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	919	(110)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	17	(2)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Seli	7,674	(919)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Seli	4,907	(282)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Self	1,893	(227)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	3,574	(429)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Self	3,107	(373)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	266	(89)	Ordinan
					2000

27-Mar-14	II IRS Securities Australia Ltd	Forter Into Derivative C	VIV	(000)	200
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	Z A N	(200)	Options
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (AVA	(175)	Options
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	N/A	78	Ontions
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	AVA	6	Ontions
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (AVA	400	Ontions
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative O	N/A	(411)	Ontions
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (ΨN	68	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	ΑN	200	Options
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (¥N	44	Options
27-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (ΑN	450	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (ΑN	375	Options
7-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (AW	(177)	Ontions
7-Mar-14	UBS Global Asset Management	Buv	2.898.280	344,022	Ordinary
7-Mar-14	UBS Global Asset Management	Buv	314,431	37,330	Ordinary
7-Mar-14	UBS Global Asset Management	Buv	70.717	8 394	Ordinary
1	UBS Global Asset Management	Вцу	78.737	3.411	Ordinary
Ł	ILIBS Global Asset Management	Riw	143 220	17,000	Ordinary
7-Mar-14	I IRC Global Assat Management	Burn	277,052	2000,71	Ordinary
•	Libe Clabal Agrat Management	, ind	500,773	72,007	Orollialy
7-Mar.14	Libe Global Asset Management	buy buy	200,000	000,045	Ordinary
7 May 1.4	I IBS Global Accet Management	bay.	070'170'7	240,000	Ordinary
7 Mar 1/	Libs Global Accet Management	oni.	6/6/1	657	Oldinary
7 May 14	UBS Constitution Authorities 144	ony Soll	40,523	4,004	Orginary
41-IPM-/	Ups securities Australia Lid	Sell	755,7	(187)	Ordinary
7 May 14	UBS Securities Australia Ltd	Sell	051,130	(357)	Ordinary
7-Mar-14	I IBC Countries Australia Ltd	liac lo	4,909	(530)	Ordinary
7-Mar-14	I IBC Counting Australia 1-4		2704	(214)	Ordinary
7-Mar-14	LIBC Countries Australia Ltd	Jel Coll	500 5	(202)	Crainary
7-Mar-14	Let Charter Australia DO CO.	130	000,0	(1000)	Ciciliary
7-Mar-14	LIBS Sequenties Australia Ltd	100	375 1	(120)	Ordinary
7-Mar-14	I BS Securities Australia I td	Soll	9010	(1,100)	Ordinary
7-Mar-14	LIBS Securities Australia Ltd	9	117,653	(13 073)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	286	(34)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	766 77	(50 6)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	[as	30 590	(3 633)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Self	16.267	(1.932)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	las	6 262	(1 100)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	les.	102.892	(12,220)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	25.706	(3.053)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	la _S	25.706	(3.053)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	38	1.996	(237)	Ordinary
7-Mar-14	UBS Securities Australia I td	وا	82 609	(9.811)	Ordinapy
7-Mar-14	I IRS Countries Australia 144	To Sol	200,20	(104)	Croining
Z-Mar-14	I IBC Countries Australia Ltd	100	0/0	(401)	Ordinary
7 1425 14	Tibe countries Australia Ltd	Jac I	750,75	(1/0)	Orginary
-Ivial - 14	Opp Securities Australia Ltd	nac n	1/4,2//	(269'07)	Ordinary
7 N. E	UBS Securities Australia Ltd	is a	17,733	(2,100)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	- Ne	10 2	(a)	Ordinary
/-Mar-14	UBS Securities Australia Ltd	Jan Jan	/05	(09)	Ordinary
/-Mar-14	UBS Securities Australia Ltd	Seli	176	(601)	Ordinary
-Mar-14	UBS Securities Australia Ltd	Sell	1,842	(218)	Ordinary
/-Mar-14	UBS Securities Australia Ltd	Sell	3,671	(435)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	2,621	(999)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	42	(2)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	1,998	(237)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	305	(107)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	8,219	(615)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	3,519	(416)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Şell	2,606	(308)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sel	4,571	(539)	Ordinary
27-Mar-14	Securities	Sell	3,123	(370)	Ordinary
7-Mar-14	UBS Securities Australia Ltd	Sell	2,591	(307)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	4,774	(292)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	17	(2)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	4,428	(524)	Ordinary
27-Mar-14	UBS Securities Australia Ltd	Sell	3,701	(438)	Ordinary
		-	7107	1001,	

20 44 00	1 10 C				The second
28-Mar-14	I IRS Securities Australia Hd	liac J	3,5/4	(420)	Ordinary
28-Mar-14	UBS Securities Australia I to	F 3	0.50	(306)	Ordings
28-Mar-14	URS Securities Australia 14d	To D	2,210	(080)	Ordinary
28-Mar-14	TIRS Sportition Australia Ind	Coll	700.0	(795)	Orginary
28-Mar-14		150	000'0	(403)	Oramary
28-Mar-14	LIRS Securities A setralia 14d	900	26.30	(100)	Cignaly
28-Mar-14	I IRC Securities Australia 14d	100	4,017	(349)	Ordinary
28-Mar-14	I IBS Securities Australia 14d	= 20	1 505	(190)	Ordinary
28-Mar-14	I IRS Securities Australia Ltd	Tool I	1,505	(180)	Ordinary
28-Mar-14	I IRS Securities Australia 14d	361	7,032	(0/0)	Ordinary
28-Mar-14	I IRC Societifies Australia Ital	liec liec	9/7	(33)	Ordinary
28-Mar-14	I BC Securities Australia 114	lioc loo	C00'1	(261)	Orginary
30-Mar-14	I IRC Securities Australia 144	lisc o	1,501	(007)	Ordinary
30-Mar-14	1 BS Securities Australia 14d	Diny	160,1	(189)	Orginary
31-Mar-14	LIRCAG Australia Branch	(ne	\$ 1.5	1 200	Orginary
31-Mar-14	I IRS AG Australia Branch	Jen	15,714	(200,001)	Crainary
31-Mar-14	I IRS Societies Australia 144	ll o	410,014	(100,000)	Ordinary
31-Mar-14	I IBS Societies Australia II-d	liac	15,054	(1,833)	Ordinary
31-Mar-14	LIBS Securities Australia Itd		777 457	102,541	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Buy	7 028	008	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Sell	158 301	(18 657)	Ordinary Ordinary
31-Mar-14	UBS Securities Australia Ltd	Buv	204.202	24 098	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Sell	87.182	(10.285)	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Sell	845,750	(100,000)	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative C	X 2	(216)	Options
31-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (ΥN	(183)	Options
31-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (MA	(100)	Options
31-Mar-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	(183)	Options
31-Mar-14	UBS Securities Australia Ltd	Buy	5,854	692	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Buy	482	22	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Sell	197,346	(23,327)	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Sell	302,310	(35,734)	Ordinary
31-Mar-14	UBS Securities Australia Ltd	Sell	20,676	(2,444)	Ordinary
01-Apr-14	UBS AG Australia Branch	Seil	17,450	(7,077)	Ordinary
01-Apr-14	UBS AG London Branch	Sell	432,040	(51,393)	Ordinary
01-Apr-14	UBS AG London Branch	Buy	433,249	51,393	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Buy	12,682	1,508	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Buy	403,920	48,012	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Sell	416,748	(49,568)	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Sell	432,040	(51,393)	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Sell	102,042	(12,138)	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Sell	77,585	(9,226)	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Buy	21,116	2,516	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Sell	167,649	(19,939)	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Buy	188,499	22,407	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	200	Options
01-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative C	N/A	(183)	Options
01-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (¥≥	7	Options
01-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative Q	NA.	13	Options
01-Apr-14	LIBS Securities Australia Ltd	Enter Into Derivative C	V.V.	30	Options
01-00-14	LIBC Waste Massacant Australia 14d	Siller Into Delivative of	NA C	21	Options
01-00-14	LIBS Wealth Management Australia Ltd	Bury	201.00	05/20	Ordinary
01-Anr-14	LIBS Securities Australia 1td	Bury	25,700	97977	Orginary
01-Anr-14	UBS Sequities Australia Ind	and V	2 414	(700)	Ordinary
01-Apr-14	LIBS Securities Australia Ltd	Seal of the seal o	748	(00)	Ordinary
01-Apr-14	UBS Securities Australia 1td	Bux	78 786	5.801	Ordinary Ardinary
01-Apr-14	UBS Securities Australia Ltd	Buy	22,625	027.5	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Buy	48.786	5.801	Ordinary
01-Apr-14	UBS Securities Australia Ltd	Buy	18,073	2.149	Ordinary
02-Apr-14	UBS AG Australia Branch	Sell	114,529	(13,652)	Ordinary
02-Apr-14	UBS AG Australia Branch	Sell	2,101,527	(250,000)	Ordinary
02-Apr-14	UBS AG Australia Branch	Setl	1,268,404	(150,000)	Ordinary
02-Apr-14	7				
	FOSS AG AUSTRAIJA BEADCH	8uv	396.291	46.857	Ordinary

UBS Securities Australia i td	9	2 091	(120)	Ordinan
JBS Securities Australia Ltd	38	2,816	(338)	Ordinary
UBS Securities Australia Ltd	Sell	491	(65)	Ordinary
UBS Securities Australia Ltd	Sell	4,770	(572)	Ordinary
ies Australia Ltd	Sell	30,004,500	(3,615,000)	Ordinary
UBS Securities Australia Ltd	Sell	4,100	(491)	Ordinary
UBS Securities Australia Ltd	Sell	2,711	(325)	Ordinary
UBS Securities Australia Ltd	la S	108	(13)	Ordinary
UBS Securities Australia I td	9	71	(20)	Ordinary
UBS AG Australia Branch	Sell	55 272	(6.612)	Ordinary
JBS AG Australia Branch	Suc.	331.587	39.610	Ordinary
JBS AG London Branch	Buv	611,409	72.911	Ordinary
andon Branch	lleS.	610.010	(77 911)	Ordinan
UBS Securities Australia Ltd	Stock borrowed	₹N	6.432	Ordinary
JBS Securities Australia Ltd	Buy	5.375	642	Ordinary
JBS Securities Australia Ltd	Buy	418	20	Ordinary
UBS Securities Australia Ltd	Buy	686,138	82.073	Ordinary
JBS Securities Australia Ltd	Sell	941,759	(112,594)	Ordinary
UBS Securities Australia Ltd	Sell	24,789	(2,962)	Ordinary
ties Australia Ltd	Buy	46,695	5,585	Ordinary
ties Australia Ltd	Sell	610,010	(72,911)	Ordinary
JBS Securities Australia Ltd	Sell	51.926	(6.206)	Ordinary
JBS Global Asset Management	Buy	510,450	60,886	Ordinary
JBS Securities Australia Ltd	Sell	970	(116)	Ordinary
Securities Australia Ltd	Sell	3,795	(454)	Ordinary
JBS Securities Australia Ltd	Biv	196,845	23 546	Ordinary
JBS Securities Australia Ltd	la _S	577	(69)	Ordinary
ties Australia I to	2 6	184 706	22,004	Codings
IBS Sportities Australia Ltd	(c)	30.180	(3.610)	Ordinary
Securities Australia I to	100	VEV	(53)	Ordings
ties Australia I to	3	603	(50)	Ordinary
JBS Securities Australia 11d	3	1 413	(169)	Ordinan
IRS Societies Australia 14d	3	3636	(514)	Ordinary
JBS Securities Australia I td	3	735	(88)	Ordinary
ities Australia I td	Sell	4 484	(537)	Ordinary
UBS Securities Australia Ltd	Sell	10.498	(1.255)	Ordinary
ties Australia I td	les.	5 295	(633)	Ordinary
UBS Securities Australia Ltd	Sell	6.483	(277)	Ordinary
ies Australia Ltd	Sell	5 235	(627)	Ordinary
UBS Securities Australia I td	Sell	451	(54)	Ordinary
IBS Societies Australia 14d	li o	723 8	(5/2)	Ordinary
trice Australia 14d	190	7250	(740)	Ordinary
I IDS Countition Australia 14d	liac liac	667,6	(450)	Ordinary
ries Australia Liu		608	(78)	Ordinary
ides Australia Ltd	- X	3,161	(3/8)	Ordinary
UBS Securities Australia Ltd	Sell	3,624	(434)	Ordinary
ities Australia Ltd	Zell	4,125	(494)	Ordinary
JBS Securities Australia Ltd	. Xell	3,843	(460)	Ordinary
JBS Securities Australia Ltd	Neil	2,891	(346)	Ordinary
rities Australia Ltd	Seil	535	(64)	Ordinary
UBS Securities Australia Ltd	Sell	144,5	(425)	Ordinary
ities Australia Ltd	Sell	2,799	(332)	Ordinary
UBS Securities Australia Ltd	Sell	2,451	(293)	Ordinary
ties Australia Ltd	Sell	2,084	(249)	Ordinary
ties Australia Ltd	Sell	2,224	(292)	Ordinary
JBS Securities Australia Ltd	Sell	1,822	(218)	Ordinary
UBS Securities Australia Ltd	Sell	2,316	(277)	Ordinary
Securities Australia Ltd	Sell	2,090	(250)	Ordinary
UBS Securities Australia Ltd	Sell	3,360	(401)	Ordinary
ies Australia Ltd	Sell	378	(45)	Ordinary
UBS Securities Australia Ltd	Sell	3,260	(386)	Ordinary
UBS Securities Australia Ltd	Sell	3,951	(472)	Ordinary
UBS Securities Australia Ltd	Sell	243	(29)	Ordinary
UBS Securities Australia Ltd	Sell	779	(63)) or city
be Authorite 14d			1111	Crainary
יכי את את מוש דינה		2.841	(339)	Ordinary

10C ACTORDOR Dropoth	•			.,,	١
T	5 5	Buy	1 224	145	Ordinary
T	-ch	Buy	282	70	Ordinary
П	tch	Buy	22	9	Ordinary
	Ę	Buy	126	15	Ordinary
	Ę.	Buy	1,534	183	Ordinary
	ıch	Buy		80	Ordinary
	ē	Buy	134	16	Ordinary
02-Apr-14 UBS AG London Branch	Ð	Buy	196	95	Ordinary
Ī	-Çı	Buy	352	42	Ordinary
32-Apr-14 UBS AG London Bra	Ę.	Buv	1.492	178	Ordinary
02-Apr-14 UBS AG London Branch	Ą	Buv	830	8	Ordinary
ı	Ę	Ruy	101	12	Ordinary Ordinary
1		Ruy	2	71-	Ordinary
02-Apr-14 LIBS AG London Branch	5 -6	and a	0 00	-	Orginary
	2	and.	34	4 (Ordinary
Ť	5 +	buy	951	6	Ordinary
Ť	51-	Buy	913	109	Ordinary
- 1	ē	Buy	80		Ordinary
02-Apr-14 UBS AG London Branch	-Ch	Buy	4,612	550	Ordinary
UZ-Apr-14 UBS AG London Brai	€.	Buy	80	-	Ordinary
- 1	Ð	Buy	25	е	Ordinary
1	Ð	Buy	88	-	Ordinary
٦	-t-	Buy	88	-	Ordinary
Ĩ	ch	Buy	698	44	Ordinary
02-Apr-14 UBS AG London Branch	to.	Buy	7,257	998	Ordinary
Ī	ch	Buy	3,805	454	Ordinary
02-Apr-14 UBS AG London Branch	-6	Buy	1,610	192	Ordinary
Г	ch	Buy	3,245	387	Ordinary
02-Apr-14 UBS AG London Branch	t)	Buy	1216	145	Ordinary
Γ	ų.	Buv	2.599	310	Ordinary
Ť	t)	Buy	1.560	186	Ordinary
02-Apr-14 UBS AG London Branch	-U	Buy	2.021	241	Ordinary
Ē	÷	Buy	866	119	Ordinary
f	t)	Buy	1.602	191	Ordinary
Ē	÷	Buy	4,058	484	Ordinary
Ť	ф	Buy	969	83	Ordinary
Ī	Ð	Buv	3.530	471	Ordinary
02-Apr-14 UBS AG London Branch	t t	Buv	777	6	Ordinary
T	ŧ	Riv	1 283	152	ording.
Ť		Buy	1115	133	Ordinary
UBS AG London Branch	5	Buy	686	1112	Ordinary
	£	Buy	2,516	300	Ordinary
Ť	to the	Buy	839	100	Ordinary
02-Apr-14 UBS AG London Branch	6	Buv	80	-	Ordinary
Ī	5	Buv	000	-	Ordinary
T	5	Bcv	000	-	Ordinary
ŀ	₽	Buy	595	12	Ordinary
ı	the state of the s	Buy	3,756	448	Ordinary
02-Apr-14 UBS AG London Branch	5	Buy	511	61	Ordinary
	-G	Buy	005'6	1,133	Ordinary
02-Apr-14 UBS AG London Brar	ъ	Buy	880	105	Ordinary
Ť	+	Buy	210	25	Ordinary
Т	5	Buy	13,618	1,625	Ordinary
Г	45	Buy	3,272	390	Ordinary
UBS AG London Branch	th.	Buy	8,385	1,000	Ordinary
Ē	÷	Виу	2,641	315	Ordinary
П	ts.	Buy	2,341	279	Ordinary
	ch	Buy	25	Э	Ordinary
-Apr-14 UBS AG London Branch	ch.	Buy	302	36	Ordinary
_	ф	Buy	00	-	Ordinary
Ī	ch.	Buy	193	23	Ordinary
UBS AG London Branch	ъ	Buy	101	12	Ordinary
	ch	Buy	210	25	Ordinary
	ъ	Buy	34	4	Ordinary
ı		,)
	ŧ	D. v.	109	- [2	Ordinary

Control Bay 882 170 Control Control Control Bay 170 Control Control Control Control Bay 170 259 Control Control Bay 170 250 Control 250 Control Control Bay 250	İ			600	105	Ordinary
URS ACE LONDON BRANCH BLAY 5,000 URS ACE LONDON BRANCH BLAY 5,000 2,500 URS ACE LONDON BRANCH BLAY 2,500 2,500 URS ACE LONDON BRANCH BLAY 2,500 3,50 URS ACE LONDON BRANCH BLAY 2,500 3,50 URS ACE LONDON BRANCH BLAY 2,500 3,50 URS ACE LONDON BRANCH BLAY 2,500 3,60 URS ACE LONDON BRANCH BLAY 2,500 3,60 URS ACE LONDON BRANCH BLAY 3,60 4,60 URS ACE LONDON BRANCH BLAY 3,60 1,70 URS ACE LONDON BRANCH BLAY 3,60 1,40 URS ACE LONDON BRANCH BLAY 3,60 1,40 URS ACE LONDON BRANCH BLAY 2,60 1,40 URS ACE LONDON BRANCH BLAY 2,70 1,12 URS ACE LONDON BRANCH BLAY 2,70 1,12 URS ACE LONDON BRANCH BLAY 2,70 1,2 URS ACE LONDON BRANCH BLAY	T	AG London Branch	Buy	788	1	(20.00
URS AG London Barneth BAY 2.75 URS AG London Barneth BAY 2.75 3.0 URS AG London Barneth BAY 2.75 3.0 URS AG London Barneth BAY 2.56 3.6 URS AG London Barneth BAY 2.5 3.6 URS AG London Barneth BAY 3.7 4.6 URS AG London Barneth BAY 3.7 4.7 URS AG London Barneth BAY 3.7 3.7 URS AG London Barneth BAY 3.7 3.2	T	A G London Branch	buy Buy	000,01		Ordinary
USE AGE (condon Barreth BLAY 2,730 35 USES AGE (condon Barreth BLAY 2,720 35 USES AGE (condon Barreth BLAY 2,720 35 USES AGE (condon Barreth BLAY 2,724 35 USES AGE (condon Barreth BLAY 2,551 445 USES AGE (condon Barreth BLAY 2,552 440 USES AGE (condon Barreth BLAY 3,374 400 USES AGE (condon Barreth BLAY 3,374 400 USES AGE (condon Barreth BLAY 3,374 400 USES AGE (condon Barreth BLAY 3,374 401 USES AGE (condon Barreth BLAY 3,374 407 USES AGE (condon Barreth BLAY 3,374 407 USES AGE (condon Barreth BLAY 3,575 1,47 USES AGE (condon Barreth BLAY 3,570 4,57 USES AGE (condon Barreth BLAY 3,57 4,57 USES AGE (condon Barreth BLAY 3,57 4,57	Ť	AG London Branch	Riv	210		Ordinary
Bis AG London Banch Buy 252 33 Bis AG London Banch Buy 2,591 359 Bis AG London Banch Buy 5,561 662 USS AG London Banch Buy 5,561 662 USS AG London Banch Buy 2,563 40 USS AG London Banch Buy 3,374 445 USS AG London Banch Buy 3,374 465 USS AG London Banch Buy 3,374 467 USS AG London Banch Buy 3,375 1,178 USS AG London Banch Buy 8,274 1,178 USS AG London Banch Buy 3,471 407 USS AG London Banch Buy 2,472 1,178 USS AG London Banch Buy 2,472 1,178 USS AG London Banch Buy 2,472 1,275 USS AG London Banch Buy 2,472 2,27 USS AG London Banch Buy 2,472 2,27 USS AG London Banch Buy 2,472 <t< td=""><td>Ť</td><td>AG London Branch</td><td>Buy</td><td>21 470</td><td></td><td>Ordinary</td></t<>	Ť	AG London Branch	Buy	21 470		Ordinary
Bis AG London Barnch Buy 2,390 356 Bis AG London Barnch Buy 2,541 662 Bis AG London Barnch Buy 2,734 445 Bis AG London Barnch Buy 3,734 445 Bis AG London Barnch Buy 3,734 405 Bis AG London Barnch Buy 3,734 405 Bis AG London Barnch Buy 3,734 405 UBS AG London Barnch Buy 8,735 1,158 UBS AG London Barnch Buy 8,735 1,158 UBS AG London Barnch Buy 6,130 7,75 UBS AG London Barnch Buy 6,130 7,75 UBS AG London Barnch Buy 6,220 1,75 UBS AG London Barnch Buy 6,220 1,75 UBS AG London Barnch Buy 7,73 4,73 UBS AG London Barnch Buy 7,73 4,73 UBS AG London Barnch Buy 7,73 4,73 UBS AG London Barnch Buy 7,73	Ť	S AG London Branch	Buv	252		Ordinary
UBS AG London Banch BJW 1,151 1,151 LBS AG London Banch BLV 5,124 6,52 LBS AG London Banch BLV 5,124 6,27 LBS AG London Banch BLV 5,124 6,27 LBS AG London Banch BLV 5,124 6,27 LBS AG London Banch BLV 3,124 6,27 LBS AG London Banch BLV 5,175 3,27 LBS AG London Banch BLV 5,175 3,27 LBS AG London Banch BLV 5,17 1,26 LBS AG London Banch BLV 5,17 1,26 LBS AG London Banch BLV 5,247 1,26 LBS AG London Banch BLV 5,247 1,26 LBS AG London Banch BLV 5,417 1,26 LBS AG London Banch BLV 5,417	Ť	S AG London Branch	Buy	2,990		Ordinary
UBS AG London Branch Buy \$ 561 662 UBS AG London Branch Buy \$ 541 662 UBS AG London Branch Buy \$ 3234 401 UBS AG London Branch Buy \$ 324 402 UBS AG London Branch Buy \$ 385 1,000 UBS AG London Branch Buy \$ 387 401 UBS AG London Branch Buy \$ 362 1,158 UBS AG London Branch Buy \$ 362 1,158 UBS AG London Branch Buy \$ 2,165 1,26 UBS AG London Branch Buy \$ 2,165 1,26 UBS AG London Branch Buy \$ 2,47 2,57 UBS AG London Branch Buy \$ 2,47 2,52 UBS AG London Branch Buy \$ 2,47 2,52 UBS AG London Branch Buy		3 AG London Branch	Buy	1,151		Ordinary
UBS AG London Banch Bluy 2.853 340 UBS AG London Banch Bluy 3.734 445 UBS AG London Banch Bluy 3.734 40 UBS AG London Banch Bluy 3.35 40 UBS AG London Banch Bluy 8,355 4,100 UBS AG London Banch Bluy 8,355 1,100 UBS AG London Banch Bluy 9,704 1,158 UBS AG London Banch Bluy 9,704 1,158 UBS AG London Banch Bluy 9,704 1,158 UBS AG London Banch Bluy 5,80 7,7 UBS AG London Banch Bluy 5,80 7,2 UBS AG London Banch Bluy 7,31 4,2 UBS AG London Banch Bluy 5,64 6,20 UBS AG London Banch Bluy 7,31 4,19 UBS AG London Banch Bluy 7,31 4,19 UBS AG London Banch Bluy 7,31 4,19 UBS AG London Banch Bluy 7,31 <td></td> <td>AG London Branch</td> <td>Buy</td> <td>5,561</td> <td></td> <td>Ordinary</td>		AG London Branch	Buy	5,561		Ordinary
UBS AG London Branch Buy 37.34 46.8 LBS AG London Branch Buy 5.37.4 46.9 LBS AG London Branch Buy 5.36.9 1,000 UBS AG London Branch Buy 8.36.9 1,000 UBS AG London Branch Buy 8.36.9 1,000 UBS AG London Branch Buy 8.36.9 1,000 UBS AG London Branch Buy 2.76.4 1,58 UBS AG London Branch Buy 2.76.7 1.44 UBS AG London Branch Buy 2.76 1.26 UBS AG London Branch Buy 2.76 2.76 UBS AG London Branch Buy		s AG London Branch	Buy	2,853		Ordinary
UBS AG London Banch By \$341 657 UBS AG London Banch Buy \$341 670 UBS AG London Banch Buy \$342 \$170 UBS AG London Banch Buy \$170 \$170 UBS AG London Banch Buy \$2,765 \$30 UBS AG London Banch Buy \$2,765 \$30 UBS AG London Banch Buy \$2,75 \$20 UBS AG London Banch Buy \$2,75 \$20 <td></td> <td>s AG London Branch</td> <td>Buy</td> <td>3,734</td> <td></td> <td>Ordinary</td>		s AG London Branch	Buy	3,734		Ordinary
UBS AG London Banch Buy 33 40 UBS AG London Banch Buy 6,286 1,000 UBS AG London Banch Buy 6,289 1,000 UBS AG London Banch Buy 9,781 1,158 UBS AG London Banch Buy 9,781 1,58 UBS AG London Banch Buy 2,419 7,6 UBS AG London Banch Buy 2,741 1,58 UBS AG London Banch Buy 2,742 1,58 UBS AG London Banch Buy 2,75 1,4 UBS AG London Banch Buy 2,72 1,4 UBS AG London Banch Buy 2,62 7,9 UBS AG London Banch Buy 2,47 2,62 UBS AG London Banch Buy 2,62 2,7 UBS AG London Banch Buy 1,50 1,2 UBS AG London Banch Buy 1,50 1,2 UBS AG London Banch Buy 2,72 2,2 UBS AG London Banch Buy 1,50 2,2		3 AG London Branch	Buy	5,341		Ordinary
UBS AG involve Branch Buy 8.385 1,000 UBS AG involve Branch Buy 8.385 1,000 UBS AG involve Branch Buy 9.937 1,138 UBS AG involve Branch Buy 2,765 330 UBS AG involve Branch Buy 2,765 330 UBS AG involve Branch Buy 6,479 1,526 UBS AG involve Branch Buy 6,479 1,526 UBS AG involve Branch Buy 6,27 1,27 UBS AG involve Branch Buy 6,27 1,27 UBS AG involve Branch Buy 6,24 6,27 UBS AG involve Branch Buy 6,24 6,25 UBS AG involve Branch Buy 7,37 8,3 UBS AG involve Branch Buy 7,37 8,3 UBS AG involve Branch Buy 7,37 8,3 UBS AG involve Branch Buy 7,31 8,3 UBS AG involve Branch Buy 7,31 8,3 UBS AG involve Branch Buy	- 1	s AG London Branch	Buy	335		Ordinary
UBS AG London Banch Buy 9,835 1,000 UBS AG London Banch Buy 9,873 1,138 UBS AG London Banch Buy 2,704 1,138 UBS AG London Banch Buy 2,704 1,138 UBS AG London Banch Buy 2,761 1,56 UBS AG London Banch Buy 2,701 1,44 UBS AG London Banch Buy 2,84 1,501 1,24 UBS AG London Banch Buy 2,84 2,84 1,501 <td></td> <td>5 AG London Branch</td> <td>Buy</td> <td>00</td> <td>-</td> <td>Ordinary</td>		5 AG London Branch	Buy	00	-	Ordinary
UBS AG London Banch Buy 9,04 1,178 UBS AG London Banch Buy 9,04 1,178 UBS AG London Banch Buy 9,04 1,178 UBS AG London Banch Buy 5,176 130 UBS AG London Banch Buy 1,000 1,44 UBS AG London Banch Buy 1,000 1,44 UBS AG London Banch Buy 2,75 1,44 UBS AG London Banch Buy 2,77 45 UBS AG London Banch Buy 2,77 45 UBS AG London Banch Buy 2,77 2,73 UBS AG London Banch Buy 2,77 2,73 UBS AG London Banch Buy 2,73 2,82 UBS AG London Banch Buy 2,73 4,93 UBS AG London Banch Buy 2,73 4,93 UBS AG London Banch Buy 4,105 1,20 UBS AG London Banch Buy 4,105 1,20 UBS AG London Banch Buy 4,105 1,2	1	5 AG London Branch	Buy	8,385		Ordinary
UBS AG London Branch Bby 9,704 1,156 UBS AG London Branch Bby 6,419 7,65 UBS AG London Branch Bby 1,511 1,81 UBS AG London Branch Bby 1,056 1,26 UBS AG London Branch Bby 3,411 1,44 UBS AG London Branch Bby 3,07 1,44 UBS AG London Branch Bby 2,472 2,93 UBS AG London Branch Bby 2,672 2,93 UBS AG London Branch Bby 2,672 2,93 UBS AG London Branch Bby 2,572 2,93 UBS AG London Branch Bby 4,150 2,93 UBS AG London Branch Bby 4,150 2,93 UBS AG London Branch Bby		s AG London Branch	Buy	9,872		Ordinary
UBS AG London Banch Bluy 6,419 766 UBS AG London Banch Bluy 6,419 766 UBS AG London Banch Buy 1,51 1,51 UBS AG London Banch Buy 1,207 1,44 UBS AG London Banch Buy 1,207 1,44 UBS AG London Banch Buy 1,207 1,44 UBS AG London Banch Buy 2,07 7 UBS AG London Banch Buy 2,07 7 UBS AG London Banch Buy 2,07 7 UBS AG London Banch Buy 2,45 2,5 UBS AG London Banch Buy 2,45 2,5 UBS AG London Banch Buy 2,45 2,5 UBS AG London Banch Buy 2,42 2,5 UBS AG London Banch Buy 2,42 2,5 UBS AG London Banch Buy 2,50 1,4 UBS AG London Banch Buy 2,50 1,2 UBS AG London Banch Buy 1,50 2,2	Ī	s AG London Branch	Buy	9,704		Ordinary
UBS AG London Banch Buy 2765 330 UBS AG London Banch Buy 2765 126 UBS AG London Banch Buy 3,411 407 UBS AG London Banch Buy 1,056 1,25 UBS AG London Banch Buy 1,055 1,4 UBS AG London Banch Buy 201 7,4 UBS AG London Banch Buy 201 7,4 UBS AG London Banch Buy 202 1,4 UBS AG London Banch Buy 2,472 2,5 UBS AG London Banch Buy 7,7 2,472 2,5 UBS AG London Banch Buy 7,5 1,0 2,2 UBS AG London Banch Buy 7,3 1,0 2,0 UBS AG London Banch Buy 7,3 1,0 2,0 UBS AG London Banch Buy 7,3 1,0 2,0 UBS AG London Banch Buy 7,3 1,0 2,0 1,0 UBS AG London Banch Buy 7,3 <th< td=""><td>Ī</td><td>AG London Branch</td><td>Buy</td><td>6,419</td><td></td><td>Ordinary</td></th<>	Ī	AG London Branch	Buy	6,419		Ordinary
UBS AG London Branch Buy 151 178 UBS AG London Branch Buy 3411 178 UBS AG London Branch Buy 3,075 1,48 UBS AG London Branch Buy 201 24 UBS AG London Branch Buy 201 24 UBS AG London Branch Buy 201 24 UBS AG London Branch Buy 662 73 UBS AG London Branch Buy 662 73 UBS AG London Branch Buy 662 73 UBS AG London Branch Buy 6,244 652 UBS AG London Branch Buy 6,264 652 UBS AG London Branch Buy 6,264 73 UBS AG London Branch Buy 6,264 731 UBS AG London Branch Buy 6,264 731 UBS AG London Branch Buy 6,264 731 UBS AG London Branch Buy 7,31 105 UBS AG London Branch Buy 7,31 105	ı	AG London Branch	Buy	2,765		Ordinary
UBS AG London Branch Bby 3411 407 UBS AG London Branch Bby 1,207 144 UBS AG London Branch Buy 59 7 UBS AG London Branch Buy 59 74 UBS AG London Branch Buy 207 42 UBS AG London Branch Buy 2,472 25 UBS AG London Branch Buy 5,467 25 UBS AG London Branch Buy 5,472 25 UBS AG London Branch Buy 6,704 190 UBS AG London Branch Buy 4,553 54 UBS AG London Branch Buy 4,553 24	- 1	AG London Branch	Buy	151		Ordinary
URS AG London Banch Buy 1,056 1,156 URS AG London Banch Buy 1,056 1,24 URS AG London Banch Buy 201 24 URS AG London Banch Buy 201 24 URS AG London Banch Buy 820 73 URS AG London Banch Buy 820 105 URS AG London Banch Buy 820 105 URS AG London Banch Buy 820 73 URS AG London Banch Buy 820 73 URS AG London Banch Buy 5,464 622 URS AG London Banch Buy 7,311 80 URS AG London Banch Buy 6,269 74 URS AG London Banch Buy 6,784 800 URS AG London Banch Buy 6,784 800 URS AG London Banch Buy 6,781 80 URS AG London Banch Buy 6,781 80 URS AG London Banch Buy 7,531 83		. AG London Branch	Buy	3,411		Ordinary
UBS AG London Branch Buy 1,207 144 UBS AG London Branch Buy 1,207 144 UBS AG London Branch Buy 201 24 UBS AG London Branch Buy 201 24 UBS AG London Branch Buy 620 75 UBS AG London Branch Buy 662 79 UBS AG London Branch Buy 662 79 UBS AG London Branch Buy 5,472 255 UBS AG London Branch Buy 5,472 255 UBS AG London Branch Buy 5,472 259 UBS AG London Branch Buy 5,247 259 UBS AG London Branch Buy 5,247 259 UBS AG London Branch Buy 5,247 250 UBS AG London Branch Buy 7,311 873 UBS AG London Branch Buy 7,311 873 UBS AG London Branch Buy 4,150 500 UBS AG London Branch Buy 4,150 500<	ı	. AG London Branch	Buy	1,056		Ordinary
UBS AG London Branch BLY 59 7 UBS AG London Branch BLY 201 24 UBS AG London Branch BLY 237 48 UBS AG London Branch BLY 237 45 UBS AG London Branch BLY 2472 79 UBS AG London Branch BLY 2472 75 UBS AG London Branch BLY 2472 75 UBS AG London Branch BLY 2472 255 UBS AG London Branch BLY 2,472 25 UBS AG London Branch BLY 1,105 13 UBS AG London Branch BLY 1,105 13 UBS AG London Branch BLY 1,105 13 UBS AG London Branch BLY 4,533 543 UBS AG London Branch BLY 4,533 543 UBS AG London Branch BLY 4,533 54 UBS AG London Branch BLY 4,533 54 UBS AG London Branch BLY 4,533 54 <td></td> <td>AG London Branch</td> <td>Buy</td> <td>1,207</td> <td></td> <td>Ordinary</td>		AG London Branch	Buy	1,207		Ordinary
UBS AGE Landon Branch Buy 201 24 UBS AGE Landon Branch Buy 737 68 UBS AGE Landon Branch Buy 620 74 UBS AGE Landon Branch Buy 662 79 UBS AGE Landon Branch Buy 662 79 UBS AGE Landon Branch Buy 2,472 265 UBS AGE Landon Branch Buy 2,472 265 UBS AGE Landon Branch Buy 2,472 265 UBS AGE Landon Branch Buy 6,269 731 UBS AGE Landon Branch Buy 6,764 803 UBS AGE Landon Branch Buy 1,105 132 UBS AGE Landon Branch Buy 4,150 803 UBS AGE Landon Branch Buy 4,150 804 UBS AGE Landon Branch Buy 4,150	- 1	. AG London Branch	Buy	59		Ordinary
UBS AG London Branch Buy 620 74 UBS AG London Branch Buy 620 74 UBS AG London Branch Buy 620 79 UBS AG London Branch Buy 675 9 UBS AG London Branch Buy 775 9 UBS AG London Branch Buy 7,87 75 UBS AG London Branch Buy 7,81 80 UBS AG London Branch Buy 7,311 180 UBS AG London Branch Buy 1,105 1.25 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 6,704 80 UBS AG London Branch Buy 6,731 7,31 12 UBS AG London Branch Buy 6,704 80 44 UBS AG London Branch Buy 6,704 80 UBS AG London Branch Buy 7,53 23 UBS AG London Branch Buy 2,707 33 UBS AG London Branch Buy <	- 1	. AG London Branch	Buy	201	24	Ordinary
UBS AG Landon Branch Buy 620 74 UBS AG Landon Branch Buy 377 65 UBS AG Landon Branch Buy 662 79 UBS AG Landon Branch Buy 7.7 79 UBS AG Landon Branch Buy 2,472 295 UBS AG Landon Branch Buy 2,472 295 UBS AG Landon Branch Buy 1,507 132 UBS AG Landon Branch Buy 1,507 132 UBS AG Landon Branch Buy 1,507 180 UBS AG Landon Branch Buy 1,507 180 UBS AG Landon Branch Buy 6,731 87 UBS AG Landon Branch Buy 4,190 500 UBS AG Landon Branch Buy 4,190 501 UBS AG Landon Branch Buy 4,190 8.2 UBS AG Landon Branch Buy 1,207 3.2 UBS AG Landon Branch Buy 2,222 2.2 UBS AG Landon Branch Buy 2,222 2.	- 1	AG London Branch	Buy	737		Ordinary
UBS AG London Branch Buy 880 45 UBS AG London Branch Buy 880 105 UBS AG London Branch Buy 662 79 UBS AG London Branch Buy 2,472 295 UBS AG London Branch Buy 5,661 622 UBS AG London Branch Buy 1,702 227 UBS AG London Branch Buy 1,7507 189 UBS AG London Branch Buy 1,7507 189 UBS AG London Branch Buy 7,311 187 UBS AG London Branch Buy 4,150 50 UBS AG London Branch Buy 4,150 50 UBS AG London Branch Buy 4,553 54 UBS AG London Branch Buy 4,553 54 UBS AG London Branch Buy 4,553 54 UBS AG London Branch Buy 2,707 32 UBS AG London Branch Buy 2,707 32 UBS AG London Branch Buy 2,534 10	1	AG London Branch	Buy	620		Ordinary
UBS AG London Branch Buy 662 79 UBS AG London Branch Buy 662 79 UBS AG London Branch Buy 2,472 295 UBS AG London Branch Buy 2,472 295 UBS AG London Branch Buy 5,626 749 UBS AG London Branch Buy 1,507 180 UBS AG London Branch Buy 1,507 180 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 4,150 5.3 UBS AG London Branch Buy 4,553 5.3 UBS AG London Branch Buy 4,553 5.3 UBS AG London Branch Buy 2,707 3.3 UBS AG London Branch Buy 2,707 3.3 UBS AG London Branch Buy 2,707 3.4 UBS AG London Branch Buy 2,851	1	AG London Branch	Buy	377		Ordinary
UBS AG London Branch Buy 75 UBS AG London Branch Buy 77 UBS AG London Branch Buy 7,464 622 UBS AG London Branch Buy 5,464 622 UBS AG London Branch Buy 1,507 132 UBS AG London Branch Buy 1,507 130 UBS AG London Branch Buy 1,507 180 UBS AG London Branch Buy 6,731 8/3 UBS AG London Branch Buy 6,731 8/3 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,190 50 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 4,553 544 UBS AG London Branch Buy 2,222 285 UBS AG London Branch Buy 1,534 183 UBS AG London Branch Buy 2,222 285 <t< td=""><td>1</td><td>AG London Branch</td><td>Buy</td><td>880</td><td></td><td>Ordinary</td></t<>	1	AG London Branch	Buy	880		Ordinary
UBS AG London Branch Buy 75 9 UBS AG London Branch Buy 2,472 295 UBS AG London Branch Buy 5,464 622 UBS AG London Branch Buy 1,302 227 UBS AG London Branch Buy 1,750 132 UBS AG London Branch Buy 1,701 187 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 4,533 544 UBS AG London Branch Buy 2,707 332 UBS AG London Branch Buy 2,534 183 UBS AG London Branch Buy 2,534 184 UBS AG London Branch Buy 2,534	- 1	AG London Branch	Buy	299	79	Ordinary
UBS AG London Branch Buy 5.472 2.95 UBS AG London Branch Buy 5.442 6.52 UBS AG London Branch Buy 1,902 7.37 UBS AG London Branch Buy 1,507 130 UBS AG London Branch Buy 7,311 800 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 4,150 5.0 UBS AG London Branch Buy 4,150 5.0 UBS AG London Branch Buy 4,53 5.3 UBS AG London Branch Buy 4,53 5.3 UBS AG London Branch Buy 4,53 5.3 UBS AG London Branch Buy 2,707 33 UBS AG London Branch Buy 2,707 33 UBS AG London Branch Buy 2,707 32 UBS AG London Branch Buy 2,707 32 UBS AG London Branch Buy 2,851	- 1	AG London Branch	Buy	75	O	Ordinary
UBS AG London Branch Buy 5,464 622 UBS AG London Branch Buy 1,502 7.37 UBS AG London Branch Buy 1,507 1.25 UBS AG London Branch Buy 1,507 1.87 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,190 50 UBS AG London Branch Buy 4,523 5.4 UBS AG London Branch Buy 4,523 5.4 UBS AG London Branch Buy 3,53 3.5 UBS AG London Branch Buy 2,222 2,222 UBS AG London Branch Buy 1,534 183 UBS AG London Branch Buy 2,222 2,222 UBS AG London Branch Buy 2,507 3.24 UBS AG London Branch Buy 2,851<	- 1	. AG London Branch	Buy	2,472	295	Ordinary
UBS AG London Branch Buy 1,902 227 UBS AG London Branch Buy 1,902 727 UBS AG London Branch Buy 1,707 180 UBS AG London Branch Buy 1,507 180 UBS AG London Branch Buy 7,311 180 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,190 50 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 3,53 54 UBS AG London Branch Buy 2,732 2,53 UBS AG London Branch Buy 2,707 3,23 UBS AG London Branch Buy 2,534 183 UBS AG London Branch Buy 2,534 18 UBS AG London Branch Buy 2,534	- 1	AG London Branch	Buy	5,464		Ordinary
UBS AG London Branch Buy 6,269 749 UBS AG London Branch Buy 1,105 1.22 UBS AG London Branch Buy 1,517 180 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 6,704 80 UBS AG London Branch Buy 4,150 5.0 UBS AG London Branch Buy 4,53 5.43 UBS AG London Branch Buy 4,53 5.43 UBS AG London Branch Buy 4,53 5.43 UBS AG London Branch Buy 2,53 5.43 UBS AG London Branch Buy 1,534 1,83 UBS AG London Branch Buy 2,707 3,75 UBS AG London Branch Buy 2,707 3,70 UBS AG London Branch Buy 2,707 3,70 UBS AG London Branch Buy 2,707 3,70 UBS AG London Branch Buy 2,851 </td <td>┪</td> <td>AG London Branch</td> <td>Buy</td> <td>1,902</td> <td></td> <td>Ordinary</td>	┪	AG London Branch	Buy	1,902		Ordinary
UBS AG London Branch Buy 1,105 132 UBS AG London Branch Buy 1,507 180 UBS AG London Branch Buy 1,504 180 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,553 5.43 UBS AG London Branch Buy 4,553 5.43 UBS AG London Branch Buy 4,553 5.4 UBS AG London Branch Buy 2,222 2.2 UBS AG London Branch Buy 1,534 183 UBS AG London Branch Buy 2,222 2.6 UBS AG London Branch Buy 1,534 183 UBS AG London Branch Buy 1,537 3.4 UBS AG London Branch Buy 2,222 2.6 UBS AG London Branch Buy 2,851 3.4 UBS AG London Branch Buy 2,851		AG London Branch	Buy	692'9		Ordinary
UBS AG London Branch Buy 1,507 180 UBS AG London Branch Buy 1,507 180 UBS AG London Branch Buy 5,704 80 UBS AG London Branch Buy 4,190 50 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 3,533 54 UBS AG London Branch Buy 3,153 54 UBS AG London Branch Buy 2,232 285 UBS AG London Branch Buy 2,153 376 UBS AG London Branch Buy 2,232 285 UBS AG London Branch Buy 2,232 40 UBS AG London Branch Buy 2,534 18 UBS AG London Branch Buy 2,534 18 UBS AG London Branch Buy 2,851		AG London Branch	Buy	1,105		Ordinary
UBS AG London Branch Buy 1,507 180 UBS AG London Branch Buy 7,311 803 UBS AG London Branch Buy 6704 800 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,533 5.43 UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 4353 5.43 UBS AG London Branch Buy 4353 35 UBS AG London Branch Buy 4353 35 UBS AG London Branch Buy 2,232 285 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 2,707 324 UBS AG London Branch Buy 2,851 40 UBS AG London Branch Buy 2,851 30 UBS AG London Branch Buy 2,851 30 UBS AG London Branch Buy 2,851	1	AG London Branch	Buy	17	2	Ordinary
UBS AG London Branch Buy 7,311 813 UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 369 44 UBS AG London Branch Buy 233 54 UBS AG London Branch Buy 193 23 UBS AG London Branch Buy 1,534 183 UBS AG London Branch Buy 2,222 265 UBS AG London Branch Buy 1,534 183 UBS AG London Branch Buy 2,707 333 UBS AG London Branch Buy 2,707 144 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 2,851 <t< td=""><td></td><td>AG London Branch</td><td>Buy</td><td>1,507</td><td>180</td><td>Ordinary</td></t<>		AG London Branch	Buy	1,507	180	Ordinary
UBS AG London Branch Buy 6,704 800 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 233 54 UBS AG London Branch Buy 453 54 UBS AG London Branch Buy 2,333 54 UBS AG London Branch Buy 3,153 54 UBS AG London Branch Buy 2,222 255 UBS AG London Branch Buy 2,707 323 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 2,534 18 UBS AG London Branch Buy 1,534 18 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1	_	AG London Branch	Buy	7,311		Ordinary
UBS AG London Branch Buy 418B 38 UBS AG London Branch Buy 4,190 500 UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 433 53 UBS AG London Branch Buy 433 53 UBS AG London Branch Buy 2,232 285 UBS AG London Branch Buy 2,707 33 UBS AG London Branch Buy 2,707 32 UBS AG London Branch Buy 2,707 32 UBS AG London Branch Buy 2,707 14 UBS AG London Branch Buy 2,851 40 UBS AG London Branch Buy 8 1	Ť	AG London Branch	Buy	6,704		Ordinary
UBS AG tenden Branch Buy 4.190 500 UBS AG tenden Branch Buy 4.190 500 UBS AG tenden Branch Buy 4.553 543 UBS AG tenden Branch Buy 369 44 UBS AG tenden Branch Buy 233 35 UBS AG tenden Branch Buy 133 35 UBS AG tenden Branch Buy 133 36 UBS AG tenden Branch Buy 2,222 265 UBS AG tenden Branch Buy 1,534 183 UBS AG tenden Branch Buy 2,222 265 UBS AG tenden Branch Buy 1,534 183 UBS AG tenden Branch Buy 2,707 333 UBS AG tenden Branch Buy 2,851 340 UBS AG tenden Branch Buy 2,851 3		AG London Branch	Buy	318		Ordinary
UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 369 44 UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 233 53 UBS AG London Branch Buy 453 54 UBS AG London Branch Buy 453 54 UBS AG London Branch Buy 2,722 23 UBS AG London Branch Buy 2,722 255 UBS AG London Branch Buy 2,707 323 UBS AG London Branch Buy 2,707 324 UBS AG London Branch Buy 2,707 324 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1		AG London Branch	Buy	4,190	200	Ordinary
UBS AG London Branch Buy 4,553 543 UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 17 2 UBS AG London Branch Buy 433 35 UBS AG London Branch Buy 433 54 UBS AG London Branch Buy 153 23 UBS AG London Branch Buy 2,222 285 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 2,851 40 UBS AG London Branch Buy 2,851 34 UBS AG London Branch Buy 8 1 UBS AG London Branch Buy 8 1 <		AG London Branch	Buy	17	2	Ordinary
UBS AG London Branch Buy 369 44 UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 293 35 UBS AG London Branch Buy 453 54 UBS AG London Branch Buy 453 54 UBS AG London Branch Buy 2,222 255 UBS AG London Branch Buy 2,727 255 UBS AG London Branch Buy 2,707 2,70 UBS AG London Branch Buy 2,707 144 UBS AG London Branch Buy 5,26 40 UBS AG London Branch Buy 5,26 14 UBS AG London Branch Buy 1,207 144 UBS AG London Branch Buy 8 1 UBS AG London Branch Buy 8 1 <t< td=""><td>Г</td><td>AG London Branch</td><td>Buy</td><td>4,553</td><td>543</td><td>Ordinary</td></t<>	Г	AG London Branch	Buy	4,553	543	Ordinary
UBS AG London Branch Buy 84 10 UBS AG London Branch Buy 17 2 UBS AG London Branch Buy 453 54 UBS AG London Branch Buy 453 54 UBS AG London Branch Buy 2,322 23 UBS AG London Branch Buy 2,722 255 UBS AG London Branch Buy 2,707 323 UBS AG London Branch Buy 2,707 324 UBS AG London Branch Buy 2,85 40 UBS AG London Branch Buy 356 40 UBS AG London Branch Buy 50 6 UBS AG London Branch Buy 8 1 UBS AG Lond		AG London Branch	Buv	369	44	Ordinary
UBS AG London Branch Buy 17 2 UBS AG London Branch Buy 433 35 UBS AG London Branch Buy 433 23 UBS AG London Branch Buy 3153 376 UBS AG London Branch Buy 2,222 265 UBS AG London Branch Buy 1,534 183 UBS AG London Branch Buy 2,707 323 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 2,851 40 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1	Γ	AG London Branch	Bir	84	Ç	Ordinan
UBS AG Landon Blanch Buy 293 35 UBS AG Landon Blanch Buy 433 54 UBS AG Landon Blanch Buy 433 36 UBS AG Landon Blanch Buy 2,222 265 UBS AG Landon Blanch Buy 2,707 33 UBS AG Landon Blanch Buy 2,707 37 UBS AG Landon Blanch Buy 2,26 27 UBS AG Landon Blanch Buy 336 40 UBS AG Landon Blanch Buy 1,207 144 UBS AG Landon Blanch Buy 8 1 UBS AG Landon	Ť	AG London Branch	Bir	17	2 ^	Ordinany
UBS AG Landon Branch Buy 453 54 UBS AG Landon Branch Buy 453 54 UBS AG Landon Branch Buy 3,153 376 UBS AG London Branch Buy 2,222 255 UBS AG London Branch Buy 2,707 323 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 326 40 UBS AG London Branch Buy 50 6 UBS AG London Branch Buy 8 1 UBS AG London Bra	T	AG London Branch	Ruy	203	35	Ordinary
UBS AG London Branch Buy 193 23 UBS AG London Branch Buy 3153 376 UBS AG London Branch Buy 2,222 376 UBS AG London Branch Buy 2,707 323 UBS AG London Branch Buy 2,707 323 UBS AG London Branch Buy 2,707 325 UBS AG London Branch Buy 356 40 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1 UBS	Т	AG London Branch	Rux	657	3 2	Ordinary
UBS AG London Branch Buy 3,153 376 UBS AG London Branch Buy 2,222 265 UBS AG London Branch Buy 2,707 333 UBS AG London Branch Buy 2,707 337 UBS AG London Branch Buy 2,26 27 UBS AG London Branch Buy 336 40 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1 UBS AG L	T	AG London Branch	Rin	103	1, 50	Ordinary
UBS AG London Branch Buy 2,722 265 UBS AG London Branch Buy 2,722 265 UBS AG London Branch Buy 2,707 323 UBS AG London Branch Buy 2,707 326 UBS AG London Branch Buy 336 40 UBS AG London Branch Buy 2,651 340 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1 UBS AG Londo	T	AG London Branch	bio.	0110	52.0	Ordinary
UBS AG London Branch Buy 1,534 20 UBS AG London Branch Buy 2,707 333 UBS AG London Branch Buy 2,707 333 UBS AG London Branch Buy 356 40 UBS AG London Branch Buy 50 6 UBS AG London Branch Buy 2,851 144 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1 UBS AG London Bran	Ť	AG London Branch	600	555 5	390	Ordinary
UBS AG London Branch Buy 2,707 313 UBS AG London Branch Buy 2,707 326 27 UBS AG London Branch Buy 2,26 27 40 UBS AG London Branch Buy 50 6 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1	T	AG London Branch	, and	777'7	507	Ordinary
UBS AG London Branch Buy 2.707 2.7 UBS AG London Branch Buy 2.26 2.5 UBS AG London Branch Buy 5.0 6 UBS AG London Branch Buy 1.207 1.4 UBS AG London Branch Buy 2.851 340 UBS AG London Branch Buy 8 1	Ť	AG London Branch	śna d	40C,1	501	Oldinary
UBS AG London Branch Buy 336 40 UBS AG London Branch Buy 336 40 UBS AG London Branch Buy 1,207 144 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1	T	AG LONION Branch	Buy	70/7	525	Ordinary
UBS AG London Branch Buy 550 40 UBS AG London Branch Buy 50 6 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1	T	AG London Branch	Buy	977	/7	Ordinary
URS AG London Branch Buy 1,207 144 UBS AG London Branch Buy 2,851 340 UBS AG London Branch Buy 8 1	Ť	AG LONGON BRAINCH	g and	3330	40	Ordinary
USE AND LONGOID BRINCH BUJ L. L	T	AG London Branch	Buy	500	9 ;	Ordinary
UBS AG London Branch Buy 8 1	T	AG LONDON BRANCH	buy.	1,207	44	Ordinary
USS AG London Branch Buy 8 1 UBS AG London Branch Buy 8 1	Ť	AG London Branch	Buy	1,857	340	Ordinary
UBS AG London Branch Buy 8 1 ULBS AG London Branch Buy 8 1 UBS AG London Branch Buy 8 1 UBS AG London Branch Buy 8 1 UBS AG London Branch Buy 8 1	Ť	AG London Branch	Buy	20		Ordinary
UBS AG London Branch Buy 8 1	1	AG London Branch	Buy	80		Ordinary
UBS AG London Branch Buy 8 1 LUBS AG London Branch Buy 8 1 LUBS AG London Branch Buy 8 1	Ť	AG London Branch	Buy	00	-	Ordinary
UBS AG London Branch Buy 8 1	Ť	AG London Branch	Buy	00	-	Ordinary
UBS AG London Branch Buy 8 1	Т	AG London Branch	Buy	8	-	Ordinary
						,

2-Apr-14	UBS AG London Branch	Buy	65,655	7,830	Ordinary
2-Apr-14	UBS AG London Branch	Buy	10,028	1,196	Ordinary
2-Apr-14	UBS AG London Branch	Buy	15,043	1,794	Ordinary
2-Apr-14	UBS AG London Branch	Buy	777,6	1,166	Ordinary
12-Apr-14	UBS AG London Branch	Buv	5.870	200	Ordinary
02-Apr-14	LIRS AG London Branch	Bir	5 203	000	
2-Anr-14	LIBS AG London Branch	8,00	20,10	300	Ordinary
12-Apr. 14	1 lbc AC London Drawah	900	0,0,0	00/	Ordinary
1	The Action Control	áng.	/hC,/	nos :	Ordinary
1-Idw-20	USS AG LOTIGOTI BIRATICII	Ang	3,102	3/0	Ordinary
02-Apr-14	UBS AG London Branch	Buy	7,547	006	Ordinary
72-Apr-14	UBS AG London Branch	Buy	12,175	1,452	Ordinary
32-Apr-14	UBS AG London Branch	Buy	66,493	7.930	Ordinary
Apr-14	UBS AG London Branch	Buy	17,231	2.055	Ordinary
Apr-14	UBS AG London Branch	Buv	9,810	1 1 20	Ordinary
02-Apr-14	UBS AG London Branch	Birv	2020	0111	or diago
Apr. 14	I BC AG London Branch	600	100,0	0 00	Ciginary
1 2	The Action of the Action	fne	6,365	000,1	Ordinary
יייייייייייייייייייייייייייייייייייייי	UBS AG LOTIGOT BRAILOT	gnà	хо	-	Ordinary
2-Apr-14	UBS At London Branch	Buy	80	-	Ordinary
32-Apr-14	UBS AG London Branch	Buy	18,615	2,220	Ordinary
-Apr-14	UBS AG London Branch	Buy	12,150	1,449	Ordinary
02-Apr-14	UBS AG London Branch	Buy	25,113	2,995	Ordinary
-Apr-14	UBS AG London Branch	Buv	29.129	3.474	Ordinary
2-Apr-14	UBS AG London Branch	Buy	27 243	3 240	Ordinary
2-Apr-14	I BS AG London Branch	Bin	230.55	275.0	in in in
2-Apr-14	I loc A C London Branch	, ind	700,62	16/2	Ordinary
11	USS ACTUAL BISHCH	bouy.	27,780	6,832	Ordinary
7 7 7 7	Cost Ad London Black	, and	685,22	7,694	Ordinary
17-Mpi-14	UBS AG LONGON Branch	any	8/9/9	795	Ordinary
-Apr-14	UBS AG London Branch	Buy	159,659	19,007	Ordinary
2-Apr-14	UBS Securities Australia Ltd	Buy	18,689	2.223	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Buy	469,404	55.964	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Seil	417 650	(40.818)	Ordinan
2-Apr-14	I IRS Securities Australia I to	100	000 (1)	(010,01)	(in in in
Δnr.14	(RC Convirties Australia 1+4		200,70	(5,00,0)	Ordinary
1	COS Securities Australia Lita	had	20,502	24,455	Ordinary
ADI-14	UBS Securities Australia Ltd	Sell	1,403	(167)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	571	(89)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Buy	3,486	415	Ordinary
Apr-14	UBS Securities Australia Ltd	Buy	54,180	6.450	Ordinary
02-Apr-14	UBS Securities Australia Ltd	loy	67	(0)	Ordinan
Apr.14	I IRS Securities Australia I to	leo.	100 200	(2)	Cicinaly
14	A DO Commission A section 14.2	Jest D-ci	66,77	(4,034)	Counary
2-Apr-14	Ubs securities Australia Ltd	Seil	9,559	(1,140)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	97,165	(11,588)	Ordinary
32-Apr-14	UBS Securities Australia Ltd	Sell	85,047	(10,141)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	83,850	(10,000)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	633 189	(75,575)	Ordinary
Anr-14	I IRS Securities Australia I to	000	484 457	(200 (2)	Ordinary.
Apr. 14	Late Constitution Authority Constitution Co.	100	101,437	(070'/5)	Cidinal y
	ODS SECULIES AUSILIANA LIU	100	3,049	(363)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	1,159	(138)	Ordinary
JZ-Apr-14	UBS Securities Australia Ltd	Sell	6,913	(823)	Ordinary
J2-Apr-14	UBS Securities Australia Ltd	Sell	2,990	(326)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	5,561	(662)	Ordinary
Apr-14	UBS Securities Australia Ltd	Cell	1 151	(197)	Carding
12-Anr-14	Constitution Assistant	150	000	(10.1)	- G
1 0 V C	The Committee Australia Line	190	077'	(041)	Ordinary
1-14	ODS SECURITIES AUSTRAINA LTG	Sell	/14	(82)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	840	(100)	Ordinary
Apr-14	UBS Securities Australia Ltd	Sell	5,233	(623)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Set	344	(41)	Ordinary
02-Apr-14	I IBC Securities Australia 114	Coll	0000	(326)	Ordina.
02-Apr-14	LIBS Sparriting Australia 144	100	750 61	(330)	Ordinary
1	ODS Securities Australia Ltd	liac li	13,927	(859,1)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	353	(42)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	9,610	(1,144)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Sell	798	(36)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	ll eS	1 495	(178)	Ordinary
2-Anr-14	I IBC Counting Australia 14d	100	000 0	(0/0)	Colding
02-Apr-14	I IBC Countition A metallia Little	100	2,330	(000)	Cidinary
	סחם שביתו וופס שתפוו מוומ בות	- 20	176,67		
				(70c'c)	Ordinary
02-Apr-14	UBS Securities Australia Ltd	Seil	714	(3,562)	Ordinary

1111
Ħ
Ш
Τ
П
Ī
T
Ι.
П
İ
Τ
Π
П
1
- 1

	UBS Securities Australia Ltd UBS Securities Australia Ltd		11,123 3,333 3,653 1,560 1,560 1,560 1,560 3,160 3,160 3,160 1,70 1,70 1,70 1,70 1,70 1,70 1,70 1,7	(403) (403) (403) (41,707) (1,500) (1,500) (1,86) (1,86) (1,86) (1,86) (1,86) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900) (1,900)	Ordinary Ordinary Ordinary Ordinary Ordinary
1	courties Australia Ltd courties Australia Ltd		39.633 39.633 1,566 1,566 1,580 3,580 3,460 3,166 4,210 107.187 4,117 1,17 1,138 6,483 6,483 6,483 1,386 6,483 1,386 6,483 1,386 6,483 1,386 6,483 1,386 6,483 1,386 6,483 1,386 6,483 1,386 6,483 6,4	(403) (4707) (1,500) (1,500) (2,000) (4,000) (4,200) (42) (376)	Ordinary Ordinary Ordinary Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	ecorrites Australia Ltd ecorrites Australia Ltd		39,633 1,680 1,6840 1,6840 2,556 3,168 3,168 1,76 1,76 4,17 1,17 1,138 6,483 6,483 1,388 1,388 1,388 1,388 2,138 3,168 3	(4,707) (1,500) (1,86) (2,000) (4,000) (4,000) (42) (9) (376)	Ordinary Ordinary Ordinary
	ecorrites Australia Ltd ecorrites Australia Ltd		1,5630 1,5640 1,5640 2,526 3,526 3,546 7,6 7,723 4,210 107,187 1,7 4,210 107,187 1,7 1,7 1,7 1,1 1,2 1,3 1,3 1,3 1,3 1,3 1,3 1,3 1,3 1,3 1,3	(1,500) (1,860) (2,000) (7) (300) (4,000) (42) (376)	Ordinary
4 1 7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	ecurities Australia Ltd ecurities Australia Ltd		1,566 1,566 1,566 3,660 3,160 3,166 1,77 4,210 107,187 4,210 1,17 1,17 1,17 1,138 6,483 6,483 1,218 1,218 1,388 1,	(186) (186) (2,000) (300) (4,000) (42) (9) (376)	Ordinary
1	ecurities Australia Lid ecurities Australia Lid		16,840 2,526 31,680 31,680 3,166 3,166 3,166 4,210 107,187 4,17 4,17 4,17 4,210 107,187 4,210 107,187 117 4,210 6,483 6,	(2,000) (1,000) (4,000) (42) (376)	2000
	courties Australia Ltd courties Australia Ltd		2,526 3,5680 3,5680 3,680 7,283 7,283 7,283 7,283 7,283 7,283 7,283 107,187 177 177 177 173 183 13,388 14,388 14,3	(7) (300) (4,000) (42) (9) (376)	Ordinary
	courties Australia Ltd courties Australia Ltd		2,526 33,680 33,48 3,166 7,283 4,210 107,187 4,117 17 17 6,483 6,483 6,483 12,138 12,138 12,138 3,316	(300) (4,000) (42) (9) (376)	Ordinary
	courties Australia Ltd courties Australia Ltd		33,680 34,84 76 7,283 4,210 107,187 4,117 117 6,483 6,483 13,388 13,388 13,388	(4,000) (42) (9) (376)	Ordinary
	courties Australia Ltd courties Australia Ltd		354 76 3.166 3.166 4.210 107.187 4.17 4.17 6.483 6.483 6.483 12.133 13.388 13.388 2.3316	(42)	Ordinary
	corrites Australia Ltd corrites Australia Ltd		3,166 7,283 7,283 4,210 107,187 4,117 6,483 6,483 12,133 12,133 13,386 2,316 3,316	(9)	Ordinary
1 1 1	ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd ecurities Australia Ltd		3,166 7,283 7,283 107,187 4,117 6,483 6,483 6,483 1,388 1,388 1,388 1,388 1,388 1,388	(376)	Ordinary
1 1 1 1 1 1 1 1 1 1	ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid ecurities Australia Lid		7,283 4,210 107,187 4,117 4,117 6,483 6,483 12,133 13,388 13,388 2,3316 3,348	(5)	Ordinary
	corrites Australia Ltd corrities Australia Ltd	Sel	4,210 107,187 4,117 6,483 6,483 12,133 13,388 13,388 13,388 13,388 13,388	(865)	Ordinary
1 1 1 1 1	eorities Australia Lid eorities Australia Lid eorities Australia Ltd		107,187 4,117 17 6,483 6,483 12,133 12,133 2,316 34	(500)	Ordinary
1 1 1 1 1 1 1	ecurites Australia Lid courlies Australia Lid curilies Australia Lid	Sell Sell Sell Sell Sell Sell Sell Sell	4,117 17 6,483 6,483 12,133 13,388 13,388 13,388 13,388 13,388	(12 730)	Critical
1 1 1 1 1 1	centities Australia Ltd centities Australia Ltd Australia Ltd centities Australia Ltd centities Australia Ltd centities Australia Ltd centities Australia Ltd centities Australia Ltd centities Australia Ltd centities Australia Ltd centities Australia Ltd	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	6,483 6,483 12,133 13,388 2,316	(489)	Ordinary
1111	cornites Australia Ltd contities Australia Ltd curtities Australia Ltd curtities Australia Ltd contities Australia Ltd		6,483 6,483 12,133 13,388 2,316 34	(2)	Ordinan
111	ecurities Australia Ltd counties Australia Ltd curities Australia Ltd	Sell Sell	6,483 12,133 13,388 2,316 34	(077)	Ordinary
	corrites Australia Ltd corrites Australia Ltd corrities Australia Ltd	Sell Sell	12,133	(077)	ording.
1	conflies Australia Ltd conflies Australia Ltd conflies Australia Ltd conflies Australia Ltd conflies Australia Ltd conflies Australia Ltd	Sell	13,388 2,316	(1 441)	Ordinary
	curities Australia Ltd curities Australia Ltd curities Australia Ltd curities Australia Ltd	Sell	2,316	(1 590)	Ordinary
02-Apr-14 UBS Sec	ecurties Australia Ltd ecurties Australia Ltd		34	(275)	Ordinary Ordinary
	ecurities Australia Ltd	ll g		(A)	Ordinary
02-Apr-14 LIBS Sec	Por irrition Australia I to	روا روا	4 210	(100)	Ordinary
		100	7 576	(300)	Ordinary
l	UBS Securities Australia Ltd	e	5,614	(999)	Ordinary
T	IBS Securities Australia I td	وا	1 071	(127)	Ordinary
T	JBS Securities Australia 14d	Sel.	1/0′-	(420)	Ordinary
1	Ourities Australia I td	3	100	(07)	Ordinary
12-Anr-14 I IRS Sev	Original Australia 14d	100	447	(01)	O'dillary
1	188 Societies Australia (44	100	/ * +	(00)	Oldinary
1	DC Contribution Americal Ltd	100	200 2	(6)	Oldinary
Ť	DO DECONTRES AUSTRALIA	100	5,233	(170)	Orginary
-	ecurities Australia Ltd	les.	091	(61)	Ordinary
02-Apr-14 Ubs sec	UDS Securities Australia Ltd	P	166,4	(264)	Ordinary
- 1	contres Australia Ltd	Jan 4	895,1	(Q8)	Ordinary
	scurities Australia Ltd	la N	843	(100)	Ordinary
T	UBS Securities Australia Ltd	Buy	2,105,000	250,000	Ordinary
T	UBS AG Australia Branch	Sell	211,312	(25,000)	Ordinary
03-Apr-14 UBS AG	UBS AG Australia Branch	Buy	3,234,668	382,370	Ordinary
	UBS AG London Branch	Buy	160	19	Ordinary
	G London Branch	Buy	624	74	Ordinary
33-Apr-14 UBS AG	JBS AG London Branch	Buy	3,085	366	Ordinary
Г	G London Branch	Buy	219	36	Ordinary
03-Apr-14 UBS AG	UBS AG London Branch	Biny	4 395	522	Ordinary
Т	IRS AG London Branch	, mark	1637	100	Ordinary
T	Clondon Branch	and a	1,017	26-	O'dillary
03-Apr-14 I IBS AG	Condon Branch	000	250	33	Ordinary
Ť	IBC ACTORDOR Branch	ond.	000	00 00	Ordinary
T	S London Branch	any	126'9	778	Ordinary
1	JBS AG London Branch	Buy	1,372	163	Ordinary
7	JBS AG London Branch	Buy	1,600	190	Ordinary
٦	JBS AG London Branch	Buy	4,999	593	Ordinary
	G London Branch	Buy	69		Ordinary
03-Apr-14 UBS AG	UBS AG London Branch	Buy	5,241	621	Ordinary
Г	UBS AG London Branch	Buv	1.038	173	Ordinary
T	UBS AG London Branch	Riv	7.CA.F.	407	Ordinary
T	IDS ACTORDOR Broad		1347		100
00 Apr 14 1 100 AC	Section Description	, ind	100,0	(3)	Orginary
Ť	Des Ad London Blanci	and a	705'6	150	Ordinary
1	London Branch	Ruy	/0/	48	Ordinary
	UBS AG London Branch	Buy	581	69	Ordinary
	5 London Branch	Buy	2,358	280	Ordinary
ر.	JBS AG London Branch	Buy	3,208	381	Ordinary
f	JBS AG London Branch	BCv	12.316	1.461	Ordinary
Ī	UBS AG London Branch	Buv	696	115	Ordinary
Τ	Clondon Branch	7.10	1000	700	i i
03-Apr-14 UBS AG	UBS AG London Branch	any	3,941	468	Ordinary

UBS Securities Australia Ltd UBS Securities Australia Ltd	10r-14	I BS Securities Australia Hd	2	2.478	(295)	Ordinary
USS Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell S	1	I IRC Convidios Australia I td	Coll	0	(+)	in in in
USS Scorribles Australials Ltd Self 1,277 1,27	pr-14	UBS Securities Australia Ltd	Sell	3 478	(414)	Ordinary
USS Scorrifies Australia Ltd Sell pr-14	UBS Securities Australia Ltd	Sell	2.478	(295)	Ordinary	
DES Securities Australia Ltd Sell 1,277 1,277 1,427	nr-14	I IRS Securities Australia I to	9	1 655	(107)	Ordinan
USS Securities Australia Ltd Sell 1,277 (17,27)	17	I IRC Societies Australia 144	le S	C.O., C	(1600)	Ordinary
USS Securities Australia Ltd Sell 1,277 1,427	17	I IBC Countries Australia 143	llo3	090.0	(967)	O'GILISI O'CILISI
USS Securities Australia Ltd Sell 17,787	1 1	LIDS Contraction Australia Ltd	Jac J	005,5	(400)	Ordinary
USS Securities Australia Ltd Sell 1,78 (2.2) UBS Securities Australia Ltd Sell 199 (2.3) UBS Securities Australia Ltd Sell 3.8 (4) UBS Securities Australia Ltd Sell 3.44 (4.2) UBS Securities Australia Ltd Sell 3.4 (4.2) UBS Securities Australia Ltd Sell 3.5 (4.2) UBS Securities Australia Ltd Sell 3.5 (4.7) UBS Securities Australia Ltd Sell 3.6 3.7 (4.7) UBS Securities Australia Ltd Sell 3.6 3.7 (4.7) UBS Securities Australia Ltd Sell 3.6 3.7 (4.7) UBS S		LIBS Countition Australia Ltd	100	7/7/	(20)	Ordinary
USS Securities Australia Ltd Sell 199 123 109 123 12	1	I IBC Countition Australia 14d	100	0,0,1	(176)	Ordinary
USS Securities Australia Ltd Sell Sell 118 (1.4) (2.5) (1	I IBS Securities Australia 14d	Coli	0 001	(60)	Ordinary
USS Scortifies Australia Ltd Sell 173 (14) UBS Scortifies Australia Ltd Sell 186 (14) UBS Scortifies Australia Ltd Sell 135 (42) UBS Scortifies Australia Ltd Sell 136 (43) UBS Scortifies Australia Ltd Sell 136 (43) UBS Scortifies Australia Ltd Sell 2,512 (42) UBS Scortifies Australia Ltd Sell 2,512 (42) UBS Scortifies Australia Ltd Sell 2,523 (43) UBS Scortifies Australia Ltd Sell 2,526 (73) UBS Scortifies Australia Ltd Sell 2,526 (73) UBS Scortifies Australia Ltd Sell 2,526 (73) UBS Scortifies Australia Ltd Sell 2,529 (7,22) UBS Scortifies Australia Ltd <td< td=""><td>1 1</td><td>1 IDC Countries Australia 144</td><td>1100</td><td>000</td><td>(22)</td><td>Clumary</td></td<>	1 1	1 IDC Countries Australia 144	1100	000	(22)	Clumary
UBS Securities Australia Ltd Sell 118 (14) UBS Securities Australia Ltd Sell 38 (14) UBS Securities Australia Ltd Sell 38 (17) UBS Securities Australia Ltd Sell 176 (19) UBS Securities Australia Ltd Sell 4,128 (19) UBS Securities Australia Ltd Sell 2,512 (19) UBS Securities Australia Ltd Sell 1,458 (10) UBS Securities Australia Ltd Sell 1,458 (17) UBS Securities Australia Ltd Sell 1,520 (10) UBS Securities Australia Ltd Sell 1,520 (10) UBS Securities Australia Ltd Sell 1,620 (1,520) UBS Securities Australia Ltd Sell 1,629 (1,520) UBS Securities Australia Ltd Sell 1,629 (1,200) UBS Securities Australia Ltd Sell 2,344 (1,720) UBS Securities Australia Ltd Sell 2,344 (1,720) UBS Securities Australia Ltd	1	Opp Securities Australia Ltd	= 1	017	(57)	Ordinary
UBS Securities Australia Ltd Sell 3,839 (4)9 UBS Securities Australia Ltd Sell 1,234 (4)2 UBS Securities Australia Ltd Sell 2,512 (299) UBS Securities Australia Ltd Sell 2,512 (299) UBS Securities Australia Ltd Sell 2,520 (299) UBS Securities Australia Ltd Sell 1,244 (410) UBS Securities Australia Ltd Sell 1,528 (473) UBS Securities Australia Ltd Sell 1,524 (473) UBS Securities Australia Ltd Sell 1,526 (1,523) UBS Securities Australia Ltd Sell 2,520 (1,533) UBS Securities Australia Ltd Sell 2,520 (1,533) UBS Securities Australia Ltd Sell 2,520 (1,533) UBS Securities Australia Ltd Sell 2,542 (1,73) UBS Securities Australia Ltd Sell 2,542 (1,73) UBS Securities Australia Ltd Sell 2,542 (1,73) UBS Securities Aust	1	Obs Securities Australia Ltd	i S	2100	(14)	Ordinary
UBS Securities Australia Ltd Sell 1.26 (1.9) UBS Securities Australia Ltd Sell 4,126 (4.9) UBS Securities Australia Ltd Sell 2,512 (6.9) UBS Securities Australia Ltd Sell 2,512 (6.9) UBS Securities Australia Ltd Sell 1,4342 (4.10) UBS Securities Australia Ltd Sell 1,4342 (4.10) UBS Securities Australia Ltd Sell 1,526 (1,529) UBS Securities Australia Ltd Sell 2,520 (1,529) UBS Securities Australia Ltd Sell 1,625 (1,529) UBS Securities Australia Ltd Sell 2,520 (1,529) UBS Securities Australia Ltd Sell 2,540 (2,591) UBS Securities Australia Ltd Sell 2,540 (3,591) UBS Sec	1	COS Securities Australia Lita	= 0	3,836	(459)	Ordinary
UBS Securities Australia Ltd Sall 133 (42) UBS Securities Australia Ltd Sell 5,512 (299) UBS Securities Australia Ltd Sell 5,512 (299) UBS Securities Australia Ltd Sell 2,512 (299) UBS Securities Australia Ltd Sell 2,512 (739) UBS Securities Australia Ltd Sell 1,244 (410) UBS Securities Australia Ltd Sell 1,258 (1,136) UBS Securities Australia Ltd Sell 2,520 (1,519) UBS Securities Australia Ltd Sell 1,059 (1,319) UBS Securities Australia Ltd Sell 2,520 (1,329) UBS Securities Australia Ltd Sell 2,520 (1,329) UBS Securitie	Pr-14	UBS Securities Australia Ltd	3	00	(i)	Ordinary
UBS Securities Australia Ltd Sell 4,126 (15) UBS Securities Australia Ltd Sell 2,512 (159) UBS Securities Australia Ltd Sell 2,512 (169) UBS Securities Australia Ltd Sell 1,4542 (410) UBS Securities Australia Ltd Sell 2,520 (1,23) UBS Securities Australia Ltd Sell 3,520 (4,10) UBS Securities Australia Ltd Sell 2,520 (1,23) UBS Securities Australia Ltd Sell 2,520 (4,13) UBS Securities Australia Ltd Sell 2,520 (1,23) UBS Securities Australia Ltd Sell 2,520 (4,23) UBS Securities Australia Ltd Sell 1,020 (1,20) UBS Securities Australia Ltd Sell 2,520 (4,27) UBS Securities Australia Ltd Sell 1,234 (4,27) UBS Securities Australia Ltd Sell 1,234 (4,27) UBS Securities Australia Ltd Sell 1,234 (4,27) UBS Securities Au	Dr-14	UBS Securities Australia Ltd	les.	353	(42)	Ordinary
UBS Securities Australia Ltd Sell 5,868 (5,736) UBS Securities Australia Ltd Sell 2,512 (790) UBS Securities Australia Ltd Sell 1,582 (1,736) UBS Securities Australia Ltd Sell 1,582 (1,736) UBS Securities Australia Ltd Sell 2,520 (300) UBS Securities Australia Ltd Sell 2,520 (1,736) UBS Securities Australia Ltd Sell 2,520 (1,736) UBS Securities Australia Ltd Sell 2,520 (1,737) UBS Securities Australia Ltd Sell 1,534 (1,738) UBS Securities Australia Ltd Sell 1,534 (1,739) UBS Securities Australia Ltd Sell 2,534 (1,739) UBS Securities Australia Ltd Sell 2,544 (1,731) UBS Securities Australia Ltd Sell 2,544 (7,731) UBS Securities Australia Ltd Sell 2,544 (7,731) UBS Securities Australia Ltd Sell 2,544 (7,731) UB	/pr-14	UBS Securities Australia Ltd	Nell Nell	971	(15)	Ordinary
UBS Securities Australia Ltd Sell 5,863 (858) UBS Securities Australia Ltd Sell 2,520 (879) UBS Securities Australia Ltd Sell 1,4,582 (879) UBS Securities Australia Ltd Sell 1,4,582 (1,730) UBS Securities Australia Ltd Sell 3,444 (410) UBS Securities Australia Ltd Sell 1,525 (1,230) UBS Securities Australia Ltd Sell 1,526 (1,513) UBS Securities Australia Ltd Sell 2,520 (1,301) UBS Securities Australia Ltd Sell 1,0,525 (1,231) UBS Securities Australia Ltd Sell 2,520 (1,301) UBS Securities Australia Ltd Sell 1,0,40 (1,301) UBS Securities Australia Ltd Sell 1,0,40 (1,23) UBS Securities Australia Ltd Sell 1,0,40 (1,23) UBS Securities Australia Ltd Sell 2,00 (1,23) UBS Securities Australia Ltd Sell 2,00 (1,23) UBS	tpr-14	UBS Securities Australia Ltd	Sell	4,124	(491)	Ordinary
UBS Securities Australia Ltd Seli 6.512 (1.736) UBS Securities Australia Ltd Seli 6.544 (1.736) UBS Securities Australia Ltd Seli 1.26 (1.736) UBS Securities Australia Ltd Seli 1.26 (1.737) UBS Securities Australia Ltd Seli 2.528 (1.737) UBS Securities Australia Ltd Seli 2.528 (1.737) UBS Securities Australia Ltd Seli 2.538 (1.737) UBS Securities Australia Ltd Seli 2.344 (1.737)	pr-14	UBS Securities Australia Ltd	Seil	5,863	(869)	Ordinary
UBS Securities Australia Ltd Sell (8.654 (1.06) UBS Securities Australia Ltd Sell 14,526 (1.736) UBS Securities Australia Ltd Sell 2,520 (1.007) UBS Securities Australia Ltd Sell 1,525 (1.537) UBS Securities Australia Ltd Sell 1,656 (1.517) UBS Securities Australia Ltd Sell 2,520 (1.007) UBS Securities Australia Ltd Sell 1,052 (1.737) UBS Securities Australia Ltd Sell 2,402 (1.737) UBS Securities Australia Ltd Sell 1,134 (1.737) UBS Securities Australia Ltd Sell 2,402 (1.737) UBS Securities Australia Ltd Sell 1,234 (1.737) UBS Securities Australia Ltd Sell 1,234 (1.737) <	pr-14	UBS Securities Australia Ltd	Sell	2,512	(299)	Ordinary
UBS Securities Australia Ltd Sell 14,582 (1,750) UBS Securities Australia Ltd Sell 3,444 (410) UBS Securities Australia Ltd Sell 2,520 (700) UBS Securities Australia Ltd Sell 1,253 (1,138) UBS Securities Australia Ltd Sell 1,6569 (4,71) UBS Securities Australia Ltd Sell 1,600 (4,73) UBS Securities Australia Ltd Sell 1,600 (4,73) UBS Securities Australia Ltd Sell 1,134 (4,73) UBS Securities Australia Ltd Sell 2,344 (4,73) UBS Securities Australia Ltd Sell 2,344 (4,73) UBS Securities Australia Ltd Sell 2,344 (7,29) UBS Securities Australia Ltd Sell 2,50 (7,29) UBS Securiti	pr-14	UBS Securities Australia Ltd	Seli	6,854	(816)	Ordinary
UBS Securities Australia Ltd Sell 3,444 (410) UBS Securities Australia Ltd Sell 1,525 (1,531) UBS Securities Australia Ltd Sell 1,525 (1,532) UBS Securities Australia Ltd Sell 1,535 (1,131) UBS Securities Australia Ltd Sell 1,536 (1,132) UBS Securities Australia Ltd Sell 1,134 (1,132) UBS Securities Australia Ltd Sell 2,340 (1,502) UBS Securities Australia Ltd Sell 2,422 (1,502)	pr-14	UBS Securities Australia Ltd	Self	14,582	(1,736)	Ordinary
UBS Securities Australia Ltd Sell 2.520 (300) UBS Securities Australia Ltd Sell 10,525 (1,153) UBS Securities Australia Ltd Sell 10,525 (1,153) UBS Securities Australia Ltd Sell 10,525 (1,118) UBS Securities Australia Ltd Sell 10,525 (1,118) UBS Securities Australia Ltd Sell 10,080 (1,118) UBS Securities Australia Ltd Sell 10,080 (1,118) UBS Securities Australia Ltd Sell 2,324 (1,29) UBS Securities Australia Ltd Sell 1,134 (1,33) UBS Securities Australia Ltd Sell 1,134 (1,33) UBS Securities Australia Ltd Sell 2,324 (2,29) UBS Securities Australia Ltd Sell 1,134 (1,33) UBS Securities Australia Ltd Sell 1,235 (1,47) UBS Securities Australia Ltd Sell 1,235 (1,47) UBS Securities Australia Ltd Sell 1,236 (1,47) UB	pr-14	UBS Securities Australia Ltd	Sell	3,444	(410)	Ordinary
UBS Securities Australia Ltd Sell (15.53) UBS Securities Australia Ltd Sell (6.71) UBS Securities Australia Ltd Sell (6.71) UBS Securities Australia Ltd Sell (6.72) UBS Securities Australia Ltd Sell (7.118) UBS Securities Australia Ltd Sell (7.132) UBS Securities Australia Ltd Sell (7.200) UBS Securities Australia Ltd Sell (7.201) UBS Securities Australia Ltd Sell (7.202) UBS Securities Australia Ltd Sell (7.204) UBS Securities Australia Ltd Sell (7.201) UBS Securities Australia Ltd Sell (7.201) UBS Securities Australia Ltd Sell (7.201) UBS Securi	pr-14	UBS Securities Australia Ltd	Sell	2,520	(300)	Ordinary
UBS Securities Australia Ltd Sell 1,0525 (1,253) UBS Securities Australia Ltd Sell 2,636 (671) UBS Securities Australia Ltd Sell 1,2530 (320) UBS Securities Australia Ltd Sell 1,030 (1,130) UBS Securities Australia Ltd Sell 1,030 (1,23) UBS Securities Australia Ltd Sell 2,520 (1,30) UBS Securities Australia Ltd Sell 2,340 (1,23) UBS Securities Australia Ltd Sell 2,344 (1,29) UBS Securities Australia Ltd Sell 2,402 (1,29) UBS Securities Australia Ltd Sell 2,402 (1,29) UBS Securities Australia Ltd Sell 1,084 (1,29) UBS Securities Australia Ltd Sell 1,284 (1,21) UBS Securities Australia Ltd Sell 1,284 (1,73) UBS Securities Australia Ltd Sell 1,284 (1,73) UBS Securities Australia Ltd Sell 1,284 (1,73) UBS Securit	pr-14	UBS Securities Australia Ltd	Sell	126	(15)	Ordinary
UBS Securities Australia Ltd Sell 5,636 (671) UBS Securities Australia Ltd Sell 2,688 (730) UBS Securities Australia Ltd Sell 1,520 (1,913) UBS Securities Australia Ltd Sell 1,520 (1,913) UBS Securities Australia Ltd Sell 1,530 (1,200) UBS Securities Australia Ltd Sell 1,34 (1,200) UBS Securities Australia Ltd Sell 2,402 (1,200) UBS Securities Australia Ltd Sell 2,402 (1,008) UBS Securities Australia Ltd Sell 2,402 (1,009) UBS Securities Australia Ltd Sell 2,140 (1,731) UBS Securities Australia Ltd Sell 2,140 (1,731) UBS Securities Australia Ltd Sell 2,140 (1,731) UBS	pr-14	UBS Securities Australia Ltd	Sell	10,525	(1.253)	Ordinary
UBS Securities Australia Ltd Sell 2,688 (320) UBS Securities Australia Ltd Sell 1,391 (1,118) UBS Securities Australia Ltd Sell 16,069 (1,303) UBS Securities Australia Ltd Sell 1,089 (1,200) UBS Securities Australia Ltd Sell 1,089 (1,200) UBS Securities Australia Ltd Sell 8,535 (1,000) UBS Securities Australia Ltd Sell 1,134 (1,331) UBS Securities Australia Ltd Sell 2,402 (329) UBS Securities Australia Ltd Sell 1,344 (1,331) UBS Securities Australia Ltd Sell 2,402 (329) UBS Securities Australia Ltd Sell 1,344 (1,331) UBS Securities Australia Ltd Sell 1,286 (1,301) UBS Securities Australia Ltd Sell 1,284 (1,301) UBS Securities Australia Ltd Sell 2,142 (1,291) UBS Securities Australia Ltd Sell 4,284 (510) UBS S	Dr-14	UBS Securities Australia Ltd	Sell	5.636	(121)	Ordinary
UBS Securities Australia Ltd Sell 9,391 (1,118) UBS Securities Australia Ltd Sell 15,059 (1,118) UBS Securities Australia Ltd Sell 2,520 (300) UBS Securities Australia Ltd Sell 2,324 (1,200) UBS Securities Australia Ltd Sell 6,37 (1,028) UBS Securities Australia Ltd Sell 2,324 (2,79) UBS Securities Australia Ltd Sell 2,344 (2,79) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 2,134 (279) UBS Securities Australia Ltd Sell 2,022 (300) UBS Securities Australia Ltd Sell 1,134 (1,7) UBS Securities Australia Ltd Sell 1,134 (1,7) UBS Securities Australia Ltd Sell 1,236 (1,7) UBS Securities Australia Ltd Sell 1,240 (1,7) UBS Securities Australia Ltd Sell 4,205 (500) UBS Securities Aus	Dr-14	UBS Securities Australia I td	وا	7 688	(320)	Ordinary
UBS Securities Australia Ltd Sell 16,069 (1,030) UBS Securities Australia Ltd Sell 3,572 (433) UBS Securities Australia Ltd Sell 2,520 (300) UBS Securities Australia Ltd Sell 6,535 (1,028) UBS Securities Australia Ltd Sell 6,535 (1,028) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 2,402 (289) UBS Securities Australia Ltd Sell 2,402 (290) UBS Securities Australia Ltd Sell 2,142 (173) UBS Securities Australia Ltd Sell 1,235 (147) UBS Securities Australia Ltd Sell 1,378 (147) UBS Securities Australia Ltd Sell 4,435 (150) UBS Securities Australia Ltd Sell 4,435 (164) UBS Securities Austra	Dr-14	ILIBS Securities Australia I td	وا	0 301	(1118)	Ordinan
UBS Securities Australia Ltd Sell 3,520 (1,20) UBS Securities Australia Ltd Sell 3,520 (1,30) UBS Securities Australia Ltd Sell 26,93 (1,00) UBS Securities Australia Ltd Sell 2,63 (1,02) UBS Securities Australia Ltd Sell 2,344 (1,02) UBS Securities Australia Ltd Sell 2,342 (2,29) UBS Securities Australia Ltd Sell 2,402 (2,39) UBS Securities Australia Ltd Sell 2,402 (2,39) UBS Securities Australia Ltd Sell 2,402 (33) UBS Securities Australia Ltd Sell 2,520 (300) UBS Securities Australia Ltd Sell 1,596 (1,90) UBS Securities Australia Ltd Sell 2,520 (300) UBS Securities Australia Ltd Sell 2,520 (300) UBS Securities Australia Ltd Sell 4,235 (500) UBS Securities Australia Ltd Sell 4,236 (500) UBS Securities Austra	pr-14	LIBS Societies Australia Ltd	100	16.050	(1013)	Ordinary
UBS Securities Australia Ltd Sell 1,220 (1,200) UBS Securities Australia Ltd Sell 1,080 (1,200) UBS Securities Australia Ltd Sell 6535 (1,020) UBS Securities Australia Ltd Sell 6535 (1,028) UBS Securities Australia Ltd Sell 2,342 (279) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 2,008 (1,731) UBS Securities Australia Ltd Sell 2,008 (1,731) UBS Securities Australia Ltd Sell 1,336 (1,731) UBS Securities Australia Ltd Sell 1,376 (1,731) UBS Securities Australia Ltd Sell 1,376 (1,731) UBS Securities Australia Ltd Sell 4,286 (30) UBS Securities Australia Ltd Sell 4,286 (30) UBS Securities		1 IDC Cocurition Australia 14d	120	0000	(000)	Cidinary
UBS Securities Australia Ltd Sell 1,373 (473) UBS Securities Australia Ltd Sell 269 (1,200) UBS Securities Australia Ltd Sell 2,67 (32) UBS Securities Australia Ltd Sell 1,134 (135) UBS Securities Australia Ltd Sell 2,344 (279) UBS Securities Australia Ltd Sell 2,402 (389) UBS Securities Australia Ltd Sell 2,402 (390) UBS Securities Australia Ltd Sell 1,134 (1,29) UBS Securities Australia Ltd Sell 1,084 (1,29) UBS Securities Australia Ltd Sell 1,084 (1,29) UBS Securities Australia Ltd Sell 1,286 (1,39) UBS Securities Australia Ltd Sell 1,284 (1,73) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,204 (1,69) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia		ODS Securities Australia Ltd	ilec i	026,2	(One)	Ordinary
UBS Securities Australia Ltd Sell 10,080 (1,200) UBS Securities Australia Ltd Sell 67 (8) UBS Securities Australia Ltd Sell 6,535 (1,028) UBS Securities Australia Ltd Sell 1,134 (1,35) UBS Securities Australia Ltd Sell 2,344 (2,79) UBS Securities Australia Ltd Sell 4,141 (402) UBS Securities Australia Ltd Sell 4,141 (402) UBS Securities Australia Ltd Sell 1,584 (1,29) UBS Securities Australia Ltd Sell 1,586 (1,90) UBS Securities Australia Ltd Sell 1,235 (1,90) UBS Securities Australia Ltd Sell 1,235 (1,71) UBS Securities Australia Ltd Sell 2,646 (1,73) UBS Securities Australia Ltd Sell 4,236 (5,00) UBS Securities Australia Ltd Sell 4,236 (5,00) UBS Securities Australia Ltd Sell 4,200 (5,00) UBS Securities Aus	<u>5</u>	UBS Securities Australia Ltd	Sell	5/6/5	(4/3)	Ordinary
UBS Securities Australia Ltd Sell 679 (89) UBS Securities Australia Ltd Sell 6,535 (1,028) UBS Securities Australia Ltd Sell 2,344 (193) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 3,377 (402) UBS Securities Australia Ltd Sell 2,024 (2,391) UBS Securities Australia Ltd Sell 2,024 (1,29) UBS Securities Australia Ltd Sell 1,236 (147) UBS Securities Australia Ltd Sell 1,378 (147) UBS Securities Australia Ltd Sell 1,378 (147) UBS Securities Australia Ltd Sell 1,378 (147) UBS Securities Australia Ltd Sell 4,235 (150) UBS Securities Australia Ltd Sell 4,235 (150) UBS Securities Australia Ltd Sell 4,235 (150) UBS Securities Australia L	pr-14	UBS Securities Australia Ltd	Sell	10,080	(1,200)	Ordinary
UBS Securities Australia Ltd Sell 6,535 (1,028) UBS Securities Australia Ltd Sell 1,334 (1,359) UBS Securities Australia Ltd Sell 2,344 (2,79) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 4,141 (402) UBS Securities Australia Ltd Sell 2,500 (402) UBS Securities Australia Ltd Sell 1,596 (190) UBS Securities Australia Ltd Sell 1,596 (190) UBS Securities Australia Ltd Sell 2,520 (190) UBS Securities Australia Ltd Sell 2,142 (153) UBS Securities Australia Ltd Sell 2,142 (153) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,285 (520) UBS Securities Australi	pr-14	UBS Securities Australia Ltd	Sell	569	(32)	Ordinary
UBS Securities Australia Ltd Sell 1,134 (1,028) UBS Securities Australia Ltd Sell 1,134 (1,031) UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 4,141 (402) UBS Securities Australia Ltd Sell 2,0064 (2,331) UBS Securities Australia Ltd Sell 2,520 (300) UBS Securities Australia Ltd Sell 1,256 (190) UBS Securities Australia Ltd Sell 1,236 (1,731) UBS Securities Australia Ltd Sell 1,236 (1,731) UBS Securities Australia Ltd Sell 4,284 (1,731) UBS Securities Australia Ltd Sell 4,284 (1,731) UBS Securities Australia Ltd Sell 4,286 (350) UBS Securities Australia Ltd Sell 4,285 (500) UBS Securities Australia Ltd Sell 4,285 (500) UBS Securities Australia Ltd Sell 4,285 (500) UBS Securitie	pr-14	UBS Securities Australia Ltd	Sell	19	(8)	Ordinary
UBS Securities Australia Ltd Sell 1,134 (139) UBS Securities Australia Ltd Sell 2,344 (279) UBS Securities Australia Ltd Sell 2,344 (279) UBS Securities Australia Ltd Sell 4,141 (493) UBS Securities Australia Ltd Sell 2,0644 (1,29) UBS Securities Australia Ltd Sell 1,084 (1,29) UBS Securities Australia Ltd Sell 1,084 (1,29) UBS Securities Australia Ltd Sell 1,084 (1,43) UBS Securities Australia Ltd Sell 1,236 (1,47) UBS Securities Australia Ltd Sell 1,284 (1,47) UBS Securities Australia Ltd Sell 1,284 (1,17) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Austr	pr-14	UBS Securities Australia Ltd	Sell	8,635	(1,028)	Ordinary
UBS Securities Australia Ltd Sell 2,344 (279) UBS Securities Australia Ltd Sell 4,402 (286) UBS Securities Australia Ltd Sell 4,141 (402) UBS Securities Australia Ltd Sell 1,064 (1,29) UBS Securities Australia Ltd Sell 1,566 (190) UBS Securities Australia Ltd Sell 1,235 (190) UBS Securities Australia Ltd Sell 1,235 (190) UBS Securities Australia Ltd Sell 2,646 (171) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,205 (500) UBS Securities Australia Ltd Sell 4,205 (500) UBS Securities Australia Lt	pr-14	UBS Securities Australia Ltd	Sell	1.134	(135)	Ordinary
UBS Securities Australia Ltd Sell 2,402 (286) UBS Securities Australia Ltd Sell 3,471 (433) UBS Securities Australia Ltd Sell 3,674 (2,391) UBS Securities Australia Ltd Sell 2,526 (300) UBS Securities Australia Ltd Sell 1,235 (147) UBS Securities Australia Ltd Sell 2,142 (147) UBS Securities Australia Ltd Sell 1,235 (147) UBS Securities Australia Ltd Sell 1,236 (147) UBS Securities Australia Ltd Sell 4,246 (1,731) UBS Securities Australia Ltd Sell 4,246 (1,731) UBS Securities Australia Ltd Sell 4,236 (500) UBS Securities Australia Ltd Sell 4,236 (500) UBS Securities Australia Ltd Sell 4,236 (500) UBS Securities Australia Ltd Sell 4,435 (16) UBS Securities Australia Ltd Sell 3,443 (16) UBS Securities Australia	pr-14	UBS Securities Australia I td	Sell	778 6	(920)	Ordinan
UBS Securities Australia Ltd Sell 4,402 UBS Securities Australia Ltd Sell 3,377 (403) UBS Securities Australia Ltd Sell 2,520 (402) UBS Securities Australia Ltd Sell 1,084 (1,23) UBS Securities Australia Ltd Sell 1,596 (1,90) UBS Securities Australia Ltd Sell 1,235 (1,90) UBS Securities Australia Ltd Sell 1,242 (1,55) UBS Securities Australia Ltd Sell 1,284 (1,731) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,335 (71) UBS Securities Australia Ltd Sell 4,336 (500) UBS Securities Australia Ltd	11	LIBO Cocurition Australia 14d	170	COA C	(500)	Ordinary
UBS Securities Australia Ltd Sell 4,141 (402) UBS Securities Australia Ltd Sell 2,0,064 (2,391) UBS Securities Australia Ltd Sell 2,0,064 (2,391) UBS Securities Australia Ltd Sell 1,596 (1,90) UBS Securities Australia Ltd Sell 1,235 (1,47) UBS Securities Australia Ltd Sell 1,245 (1,731) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,285 (528) UBS Securities Australia Ltd Sell 4,285 (520) UBS Securities Australia Ltd Sell 4,285 (520) UBS Securities Australia Ltd Sell 4,285 (520) UBS Securities Australia Ltd Sell 3,994 (416) UBS Securities A	1	ODS Secondes Australia Liu	liac	2,402	(007)	Ordinary
UBS Securities Australia Ltd Sell 3,377 (402) UBS Securities Australia Ltd Sell 2,0,684 (1,29) UBS Securities Australia Ltd Sell 1,596 (190) UBS Securities Australia Ltd Sell 1,235 (147) UBS Securities Australia Ltd Sell 1,235 (147) UBS Securities Australia Ltd Sell 1,4540 (1,731) UBS Securities Australia Ltd Sell 2,424 (510) UBS Securities Australia Ltd Sell 4,224 (510) UBS Securities Australia Ltd Sell 4,224 (510) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 3,494 (16) UBS Securities Australia Ltd Sell 3,736 (16) UBS Securities Australia	D14	UBS Securities Australia Ltd	Sell	4,141	(493)	Ordinary
UBS Securities Australia Ltd Sell 1,0684 (2,391) UBS Securities Australia Ltd Sell 1,0684 (1,29) UBS Securities Australia Ltd Sell 1,596 (190) UBS Securities Australia Ltd Sell 1,235 (147) UBS Securities Australia Ltd Sell 2,742 (255) UBS Securities Australia Ltd Sell 2,646 (1,731) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 3,444 (1,1) UBS Securities Australia Ltd Sell 3,444 (1,6) UBS Securities Australia Ltd Sell 3,736 (1,6) UBS Securities Austra	pr-14	UBS Securities Australia Ltd	- Sel	3,377	(402)	Ordinary
UBS Securities Australia Ltd Sell 1,084 (129) UBS Securities Australia Ltd Sell 2,520 (300) UBS Securities Australia Ltd Sell 1,235 (47) UBS Securities Australia Ltd Sell 2,142 (752) UBS Securities Australia Ltd Sell 2,142 (7,731) UBS Securities Australia Ltd Sell 4,244 (510) UBS Securities Australia Ltd Sell 4,236 (164) UBS Securities Australia Ltd Sell 4,236 (500) UBS Securities Australia Ltd Sell 4,435 (16) UBS Securities Australia Ltd Sell 8 (17) UBS Securities Australia Ltd Sell 3,494 (416) UBS Securities Australia Ltd Sell 7,736 (517) UBS Securities Australia Ltd	pr-14	UBS Securities Australia Ltd	Sell	20,084	(2,391)	Ordinary
UBS Securities Australia Ltd Sell 1,556 (300) UBS Securities Australia Ltd Sell 1,596 (1490) UBS Securities Australia Ltd Sell 1,232 (4751) UBS Securities Australia Ltd Sell 2,142 (2552) UBS Securities Australia Ltd Sell 2,142 (7531) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 4,200 (50) UBS Securities Australia Ltd Sell 4,200 (50) UBS Securities Australia Ltd Sell 3,494 (16) UBS Securities Australia Ltd Sell 3,494 (16) UBS Securities Australia Ltd Sell 3,736 (21) UBS Securities Australia Ltd<	pr-14	UBS Securities Australia Ltd	Sell	1,084	(129)	Ordinary
UBS Securities Australia Ltd Sell 1,596 (190) UBS Securities Australia Ltd Sell 1,235 (147) UBS Securities Australia Ltd Sell 1,4,540 (1,731) UBS Securities Australia Ltd Sell 1,4,540 (1,731) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (500) UBS Securities Australia Ltd Sell 4,235 (500) UBS Securities Australia Ltd Sell 4,235 (500) UBS Securities Australia Ltd Sell 4,435 (500) UBS Securities Australia Ltd Sell 4,435 (500) UBS Securities Australia Ltd Sell 4,435 (7) UBS Securities Australia Ltd Sell 3,444 (1) UBS Securities Australia Ltd Sell 3,443 (1) UBS Securities Australia Ltd Sell 3,444 (1) UBS Securities Australia Ltd Sell 3,444 (1) UBS Securities Australia Ltd </td <td>or-14</td> <td>UBS Securities Australia Ltd</td> <td>laS</td> <td>2 520</td> <td>(300)</td> <td>Ordinary</td>	or-14	UBS Securities Australia Ltd	laS	2 520	(300)	Ordinary
UBS Securities Australia Ltd Seil 1,335 (147) UBS Securities Australia Ltd Seil 1,42 (255) UBS Securities Australia Ltd Seil 1,424 (1,731) UBS Securities Australia Ltd Seil 2,646 (1,731) UBS Securities Australia Ltd Seil 4,284 (510) UBS Securities Australia Ltd Seil 4,203 (500) UBS Securities Australia Ltd Seil 4,203 (7) UBS Securities Australia Ltd Seil 8 (1) UBS Securities Australia Ltd Seil 3,494 (416) UBS Securities Australia Ltd Seil 1,736 (201) UBS Securities Australia Ltd Seil 1,736 (201) UBS Securities Australia Ltd	nr-14	LiBS Securities Australia Ltd	la's	1 596	(190)	Ordinary
UBS Securities Australia Ltd Sell 1,432 (1,431) UBS Securities Australia Ltd Sell 1,434 (1,731) UBS Securities Australia Ltd Sell 2,646 (1,731) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 2,60 (500) UBS Securities Australia Ltd Sell 3,60 (6) UBS Securities Australia Ltd Sell 3,73 (1,5) UBS Securities Australia Ltd Sell 3,73 (1,5) UBS Securities Australia Ltd Sell 1,73 (1,6) UBS Securities Australia Ltd Sell 1,310 (1,6) UBS Securities Australia Ltd<	1	I IBC Countition Australia 14d	100	700,1	(57)	Ordinary
UBS Securities Australia Ltd Sell 1,142 (1,23) UBS Securities Australia Ltd Sell 2,646 (315) UBS Securities Australia Ltd Sell 1,274 (510) UBS Securities Australia Ltd Sell 1,378 (164) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,323 (500) UBS Securities Australia Ltd Sell 2,69 (32) UBS Securities Australia Ltd Sell 5,60 (500) UBS Securities Australia Ltd Sell 5,60 (416) UBS Securities Australia Ltd Sell 5,70 (500) UBS Securities Australia Ltd Sell 5,73 (15) UBS Securities Australia Ltd Sell 1,73 (15) UBS Securities Australia Ltd Sell 1,73 (1,04) UBS Securities Australia Ltd Sell 1,73 (1,04) UBS Securities Australia Ltd		ODS Securities Australia Liu	100	57,1	(14/)	Ordinary
UBS Securities Australia Ltd Seli 1,540 (1,731) UBS Securities Australia Ltd Sell 2,646 (315) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 8,0 (6) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 1,3 (16) UBS Securities Australia Ltd Sell 1,3 (16) UBS Securities Australia Ltd Sell 1,3 (17) UBS Securities Australia Ltd <td< td=""><td>pr-↓4</td><td>UBS Securities Australia Ltd</td><td>Seil</td><td>2,142</td><td>(255)</td><td>Ordinary</td></td<>	pr-↓4	UBS Securities Australia Ltd	Seil	2,142	(255)	Ordinary
UBS Securities Australia Ltd Sell 2,646 (315) UBS Securities Australia Ltd Sell 4,284 (150) UBS Securities Australia Ltd Sell 4,284 (164) UBS Securities Australia Ltd Sell 4,205 (500) UBS Securities Australia Ltd Sell 4,205 (500) UBS Securities Australia Ltd Sell 2,00 (500) UBS Securities Australia Ltd Sell 2,00 (500) UBS Securities Australia Ltd Sell 3,94 (416) UBS Securities Australia Ltd Sell 8 (17) UBS Securities Australia Ltd Sell 8 (17) UBS Securities Australia Ltd Sell 8 (17) UBS Securities Australia Ltd Sell 1,736 (17) UBS Securities Australia Ltd Sell 1,736 (21) UBS Securities Australia Ltd Sell 7,736 (201) UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Se	pr-14	UBS Securities Australia Ltd	Seli	14,540	(1,731)	Ordinary
UBS Securities Australia Ltd Sell 4,284 (510) UBS Securities Australia Ltd Sell 1,378 (164) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 269 (32) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 176 (2) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 8/744 <td>pr-14</td> <td>UBS Securities Australia Ltd</td> <td>lles:</td> <td>2,646</td> <td>(315)</td> <td>Ordinary</td>	pr-14	UBS Securities Australia Ltd	lles:	2,646	(315)	Ordinary
UBS Securities Australia Ltd Sell 1,378 (164) UBS Securities Australia Ltd Sell 412 (49) UBS Securities Australia Ltd Sell 4,435 (500) UBS Securities Australia Ltd Sell 4,435 (528) UBS Securities Australia Ltd Sell 2,00 (500) UBS Securities Australia Ltd Sell 143 (17) UBS Securities Australia Ltd Sell 3,494 (416) UBS Securities Australia Ltd Sell 3,494 (416) UBS Securities Australia Ltd Sell 8 (17) UBS Securities Australia Ltd Sell 8 (17) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 1,736 (921) UBS Securities Australia Ltd Sell 1,736 (921) UBS Securities Australia Ltd Sell 8,744 (174) UBS Securities Australia Ltd Sell 8,744 (174) UBS Securities Australia Ltd <td< td=""><td>pr-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>4.284</td><td>(510)</td><td>Ordinary</td></td<>	pr-14	UBS Securities Australia Ltd	Sell	4.284	(510)	Ordinary
UBS Securities Australia Ltd Sell 412 (49) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,350 (500) UBS Securities Australia Ltd Sell 2,69 (500) UBS Securities Australia Ltd Sell 2,69 (77) UBS Securities Australia Ltd Sell 5,60 (6) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 1,736 (9,21) UBS Securities Australia Ltd Sell 8,744 (1,04) UBS Securities Australia Ltd <t< td=""><td>nr-14</td><td>UBS Securities Australia I td</td><td>lleS</td><td>1 378</td><td>(164)</td><td>Ordinary</td></t<>	nr-14	UBS Securities Australia I td	lleS	1 378	(164)	Ordinary
UBS Securities Australia Ltd Sell 41.2 (1.9) UBS Securities Australia Ltd Sell 4.250 (500) UBS Securities Australia Ltd Sell 4.250 (500) UBS Securities Australia Ltd Sell 1.43 (77) UBS Securities Australia Ltd Sell 50 (32) UBS Securities Australia Ltd Sell 8.0 (6) UBS Securities Australia Ltd Sell 8.0 (6) UBS Securities Australia Ltd Sell 8.0 (16) UBS Securities Australia Ltd Sell 1.310 (15) UBS Securities Australia Ltd Sell 1.310 (16) UBS Securities Australia Ltd Sell 1.310 (16) UBS Securities Australia Ltd Sell 7,736 (92) UBS Securities Australia Ltd Sell 8.744 (17) UBS Securities Australia Ltd Sell 8.744 (17) UBS Securities Australia Ltd Sell 8.744 (194) UBS Securities Australia Ltd Sell </td <td></td> <td>Libo Constitution Australia 144</td> <td>-</td> <td>743</td> <td>(50)</td> <td></td>		Libo Constitution Australia 144	-	743	(50)	
UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,203 (500) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 269 (32) UBS Securities Australia Ltd Sell 50 (6) UBS Securities Australia Ltd Sell 3,494 (416) UBS Securities Australia Ltd Sell 175 (71) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Sell 8,744 (1,60) UBS Securities Australia Ltd Sell 8,744 (1,04)	į .	Opp Decar rice Australia Liu	1100	714	(64)	Clulialy
UBS Securities Australia Ltd Sell 4,435 (528) UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 269 (32) UBS Securities Australia Ltd Sell 143 (7) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 176 (2) UBS Securities Australia Ltd Sell 176 (1) UBS Securities Australia Ltd Sell 1310 (156) UBS Securities Australia Ltd Sell 1310 (156) UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Sell 8/744 (104) UBS Securities Australia Ltd Sell	pr-14	UBS Securities Australia Ltd	Sell	4,203	(200)	Ordinary
UBS Securities Australia Ltd Sell 4,200 (500) UBS Securities Australia Ltd Sell 143 (17) UBS Securities Australia Ltd Sell 50 (6) UBS Securities Australia Ltd Sell 3,494 (416) UBS Securities Australia Ltd Sell 1,75 (71) UBS Securities Australia Ltd Sell 1,75 (15) UBS Securities Australia Ltd Sell 1,73 (156) UBS Securities Australia Ltd Sell 7,73 (921) UBS Securities Australia Ltd Sell 8,74 (104)	pr-14	UBS Securities Australia Ltd	Sell	4,435	(228)	Ordinary
UBS Securities Australia Ltd Sell 269 (32) UBS Securities Australia Ltd Sell 56 (6) UBS Securities Australia Ltd Sell 56 (6) UBS Securities Australia Ltd Sell 3,494 (416) UBS Securities Australia Ltd Sell 3,694 (416) UBS Securities Australia Ltd Sell 176 (21) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Sell 8,744 (1,041)	pr-14	UBS Securities Australia Ltd	Sell	4,200	(200)	Ordinary
UBS Securities Australia Ltd Sell 143 (17) UBS Securities Australia Ltd Sell 3,494 (416) UBS Securities Australia Ltd Sell 3,494 (416) UBS Securities Australia Ltd Sell 176 (21) UBS Securities Australia Ltd Sell 1,7310 (156) UBS Securities Australia Ltd Sell 1,7316 (156) UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Sell 7,736 (104) UBS Securities Australia Ltd Sell 8,744 (1,04) UBS Securities Australia Ltd Sell 8,744 (1,04) UBS Securities Australia Ltd Sell 8,744 (1,04) UBS Securities Australia Ltd Sell (1,041)	pr-14	UBS Securities Australia Ltd	lles	996	(32)	Ordinary
UBS Securities Australia Ltd Sell 114	I IRS Sequities Australia I td	Sel	143	(17)	Ordinany	
UBS Securities Australia Ltd Sell 3.9 (16) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 176 (21) UBS Securities Australia Ltd Sell 176 (21) UBS Securities Australia Ltd Sell 1,7,36 (921) UBS Securities Australia Ltd Sell 8,744 (1,041)		The second was a second	100	C#1	(/)	Ordinary
UBS Securities Australia Ltd Sell 8 (17b) UBS Securities Australia Ltd Sell 176 (21) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 8 (921) UBS Securities Australia Ltd Sell 8 (11) UBS Securities Australia Ltd Sell 8 (24) UBS Securities Australia Ltd Sell 8 (14) UBS Securities Australia Ltd Sell 8 (24) 1043 Securities Australia Ltd Sell 8 (24) 105372 (1,949)	4	UBS Securities Australia Ltd	- T	200	(a)	Ordinary
UBS Securities Australia Ltd Sell 176 (71) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Sell 8,744 (1,041) UBS Securities Australia Ltd Sell 8,744 (1,041) UBS Securities Australia Ltd Sell (5,372 (1,949)	4-15	UDS Securities Australia Liu	Sell	3,494	(416)	Ordinary
UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 1,310 (156) UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Sell 8,744 (1,041) UBS Securities Australia Ltd Sell 8,744 (1,041) UBS Securities Australia Ltd Sell 8,744 (1,041)	or-14	IUBS Securities Australia Ltd	Sell	80	Ξ	Ordinary
UBS Securities Australia Ltd Sell 7,310 (156) UBS Securities Australia Ltd Sell 7,336 (921) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 8,744 (1,041) UBS Securities Australia Ltd Sell (6,372 (1,949)	or-14	UBS Securities Australia Ltd	Sell	176	(21)	Ordinary
UBS Securities Australia Ltd Sell 7,736 (921) UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd 8,744 (1,041) UBS Securities Australia Ltd Sell (6,372 (1,949)	or-14	UBS Securities Australia Ltd	Sell	1,310	(126)	Ordinary
UBS Securities Australia Ltd Sell 8 (1) UBS Securities Australia Ltd Sell 8,744 (1,041) UBS Securities Australia Ltd Sell 16,372 (1,949)	or-14	UBS Securities Australia Ltd	Sel	7.736	(921)	Ordinary
USS Securities Australia Ltd Seli 8,744 (1,041) UBS Securities Australia Ltd Seli 1,6372 (1,949)	14	LIBS Sequirities Australia Itd	9	α	(1)	Ordinan
UBS Securities Australia Ltd Self 16,372 (1,949)	17	I BS Sequenties Australia I to	3	0 4 4 0	(1001)	Ordinary
UBS Securities Australia Ltd Self (1,949)		ODS SECULIES AUSTRAIR LIG	200	1,744	10,1	Cluiriary
	P-14	UBS Securities Australia Ltd	Seil	17/11	2000	

03-Apr-14				-	
03-Apr-14	UBS AG London Branch	Buy	1,397	166	Ordinary
41-144-CO	UBS AG London branch	Buy	659,9	68/	Ordinary
03-Apr-14	UBS AG LONGON Branch	eny	99/	5	Ordinary
4Dr-14	UBS AG London Branch	Buy	12,396	1,474	Ordinary
4Dr-14	UBS AG London Branch	Buy	6,732	801	Ordinary
03-Apr-14	UBS AG London Branch	Buy	749	88	Ordinary
4Dr-14	UBS AG London Branch	Buy	252	30	Ordinary
Apr-14	UBS AG London Branch	Buy	673	08	Ordinary
03-Apr-14	UBS AG London Branch	Buy	294	35	Ordinary
- Id	USS AG LONDON BIANCH	BUy	2,414	/87	Ordinary
03-Apr-14	UBS AG London Branch	Buy	244	29	Ordinary
03-Apr-14	UBS AG London Branch	BUy	186,1	388	Ordinary
03-Apr-14	UBS AG London Branch	Buy	143		Ordinary
03-Apr-14	UBS AG LONGON Branch	Buy	202	24	Ordinary
Apr-14	UBS AG London Branch	Buy	25	n	Ordinary
4Dr-14	UBS AG London Branch	Buy	80	-	Ordinary
3-Apr-14	UBS AG London Branch	Buy	17	2	Ordinary
3-Apr-14	UBS AG London Branch	Buy	185	22	Ordinary
Apr-14	UBS AG London Branch	Buy	2,280	271	Ordinary
03-Apr-14	UBS AG London Branch	Buy	573	6B	Ordinary
Apr-14	UBS AG London Branch	Buy	1,978	235	Ordinary
Apr-14	UBS AG London Branch	Buy	1,238	147	Ordinary
3-Apr-14	UBS AG London Branch	Buy	2,516	299	Ordinary
J3-Apr-14	UBS AG London Branch	Buy	362	43	Ordinary
Apr-14	UBS AG London Branch	Buy	101	12	Ordinary
03-Apr-14	UBS AG London Branch	Buy	6,002	712	Ordinary
Apr-14	UBS AG London Branch	Buy	6,296	746	Ordinary
Apr-14	UBS AG London Branch	Buy	2,711	321	Ordinary
3-Apr-14	UBS AG London Branch	Buy	10,266	1,212	Ordinary
3-Apr-14	UBS AG London Branch	Buy	8,470	1,000	Ordinary
Apr-14	UBS AG London Branch	Buy	8,470	1,000	Ordinary
Apr-14	UBS AG London Branch	Buy	652	77	Ordinary
3-Apr-14	UBS AG London Branch	Buy	534	69	Ordinary
Apr-14	UBS AG London Branch	Buy	8,216	976	Ordinary
33-Apr-14	UBS AG London Branch	Buy	7,615	668	Ordinary
03-Apr-14	UBS AG London Branch	Buy	3,905	461	Ordinary
03-Apr-14	UBS AG London Branch	Buv	6.814	805	Ordinary
13-Apr-14	UBS AG tondon Branch	Buv	786.7	943	Ordinary
Apr-14	UBS AG London Branch	Buy	oc.	-	Ordinary
3-Apr-14	UBS AG London Branch	Biv	977	. E	Ordinary
33-Apr-14	UBS AG London Branch	Birv	330	000	Ording
13-Apre-14	I BC AG London Branch	600	1 314	200	Ordinary
2 Apr 14	Libe AC London Brongle	edy	1,744	907	Ordinary
1 2 2	UBS AG LOTION BIAIRI	Buy	579'1	761	Ordinary
13-Apr-14	UBS AG London Branch	Buy	1,456	172	Ordinary
33-Apr-14	UBS AG London Branch	Buy	414	49	Ordinary
33-Apr-14	UBS AG London Branch	Buy	6/8	104	Ordinary
03-Apr-14	UBS AG London Branch	Buy	719	85	Ordinary
3-Apr-14	UBS AG London Branch	Buy	968	106	Ordinary
3-Apr-14	UBS AG London Branch	Buy	761	96	Ordinary
03-Apr-14	UBS AG London Branch	Buv	1,200	147	Ordinary
03-Apr-14	UBS AG London Branch	Buv	169	202	Ordinary
03-Apr-14	UBS AG London Branch	Biv	565	202	Ordinary
03-Apr-14	UBS AG London Branch	Buv	696	114	Ordinary
Apr-14	UBS AG London Branch	Buy	912	108	Ordinary
3-Apr-14	U V	Bin	186	200	Ordinary.
03-Apr-14	LIBS AG London Branch	(S)	92	77	Ordinary
3-Anr-14	LIBS AG London Branch	Bin	1 157	765	Sediony Sediony
03-Anr-14	I IBS A G London Branch	Bury	355	50	Ordinary
03-Apr-14	PS AG London Branch	Bus	0001	100	Calibro
3-Apr-14	LIBS AG London Branch	600	060,1	677	Ordinary
03 Apr 14	Cos Ad Coldon Branch	sng.	4,200	300	Ordinary
1	List AC London Blanch	ng	010,0	451	Ordinary
03-Apr-14	UBS AG LONGON Branch	Buy	788	34	Ordinary
03-Apr-14	UBS At London Branch	Buy	102	12	Ordinary
03-Apr-14	UBS AG London Branch	Buy	8.639	1020	Ordina.
				2124	Ciciliary
03-Apr-14	UBS AG London Branch	Buy	1,220	144	Ordinary

<u></u>			<u> </u>	<u>}</u>	2 2	<u></u>	2	. 2	2	ابر	ζ.	2	ź	2	2	2	2 2	. 2	2	۲.	2	2	2	2 2	T .	2	2	2	2 2		<u></u>	2	<u> </u>	2 2	7	2 2	2	5	2	2 2	,	5	7	8	2	<u></u>	_>	Ţ,	, ,	7	у	7	,		_	>
Ц	Ordinary	_		Ordinary	1			L	Ordina	Ordina	Ordina	Ordinary					Ordinary			Ц		Ordinary			Ordinary				l		Ш		Ordinary				Ц		Ordinary		Ordinary	Ordinar	Ordinary	Ordinary	Ordinary	Ordinary	Ordinar	Ordinary								
445	1,585		464	177	35	144	1 068		-	2	136	2	42	143	1,481	1,595	80	13	150	154	587	102	430	458	205	7	61	171	Ę.	589	191	888	151	<u>α</u>	2	-	35	159	80/	77	805	64	115	17	סןי	- 8	658	924	88	1,020	254	141	145	540	1,000	246
3,747	13,330	8 000	3,900	1,088	13011	1 210	8.971	8	8	11	1,142	17	353	1,200	12,426	13,382	15,305	109	1,258	1,291	4,925	855	0,040	3.840	1,722	59	512	1,437	500,1	4,948	1,352	7,543	1,268	470	17	8	294	1,334	6,7/5	606	99/9	538	296	143	9/6	778	5,530	7,766	739	8,558	2,131	1,183	1171	4,531	8,400	990.6
Buy	Any B	Buy	suy o'''	Buy	Puy Buw	Buy	Buv	Buy	Buv	Buy	Buy	Buy	Buy	Buy	Pauly Barry	Buv	Buy	Buy	Buy	Buy	Buv	Buy	Виу	Buy	Buy	Buv	Buy	Buy	Buy	Buy	euy B.::	Riv	Buy	Buy	Buy	Buy	Buy	Buy	Brit	Buy	Buy	Buy														
UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	LIBS AG LONGOI BIGING	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	LIRS AG London Branch	UBS AG London Branch	USA AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch																											
03-Apr-14	03-Apr-14	03-Apr-14	03-Apr-14	03-Apr-14	ı	1	1	03-Apr-14	03-Apr-14	03-Apr-14		- 1	ı	H	03-Apr-14	03-Apr-14	03-Apr-14	03-Apr-14	03-Apr-14	03-Apr-14	- 1	- 1	03-Apr-14	1	03-Apr-14	03-Apr-14	03-Apr-14	- 1	03-Apr-14	1 3	1 1	03-Apr-14	03-Apr-+4	03-Apr-14																						

One State Control Bankh Byty 138 14 One State Control Bankh Byty 2375 238 Ules Ad London Bankh Byty 2,775 328 Ules Ad London Bankh Byty 2,775 328 Ules Ad London Bankh Byty 2,135 26,23 Ules Ad London Bankh Byty 2,135 1000 Ules Ad London Bankh Byty 2,135 100 Ules Ad London Bankh Byty 2,135 100 Ules Ad London Bankh Byty 2,135 100 Ules Ad London Bankh Byty 2,135 10 Ules Ad London Bankh Byty 2,136 1,13 Ules Ad London Bankh Byty 2,136 3,13 Ules Ad London Bankh Byty 2,134 3,13 Ules Ad London Bankh Byty 2,144 3,14 Ules Ad London Bankh Byty 2,124 1,17 Ules Ad London Bankh Byty 2,14 1,13 Ules Ad London Bankh Byty <th></th> <th>Buy Buy Buy Buy Buy Buy Buy Buy Buy</th> <th>237 2,775 169</th> <th>28 328 20 20 679</th> <th>Ordinary Ordinary Ordinary</th>		Buy Buy Buy Buy Buy Buy Buy Buy Buy	237 2,775 169	28 328 20 20 679	Ordinary Ordinary Ordinary
Bis AG (corder) Branch By 2,775 3.28 Bis SAG (corder) Branch Buy 5,238 6,29 USS AG (corder) Branch Buy 1,099 130 USS AG (corder) Branch Buy 2,594 2,78 USS AG (corder) Branch Buy 2,594 2,594 USS AG (corder) Branc		Buy Buy Buy Buy Buy Buy Buy	2,775	328	Ordinary
Bis AG London Banch Bluy 5,125 6,22 Bis AG London Banch Bluy 5,135 6,22 Bis AG London Banch Bluy 5,135 6,17 Bis AG London Banch Bluy 5,135 6,17 Bis AG London Banch Bluy 1,039 1,039 Bis AG London Banch Bluy 1,039 1,13 Bis AG London Banch Bluy 1,039 1,13 Bis AG London Banch Bluy 2,042 2,04 Bis AG London Banch Bluy 2,042 2,04 Bis AG London Banch Bluy 2,04 2,04 Bis AG London Banch Bluy 2,04 2,04 Bis AG London Banch Bluy 2,04 2,04 Bis AG London Banch Bluy		Buy Buy Buy Buy Buy Buy	1691	20	Ordinary
Bis AG London Barch Bly 5.32B 627 Bis AG London Barch Bly 5.32B 627 Bis AG London Barch Bly 5.13B 607 Bis AG London Barch Bly 5.13B 607 Bis AG London Barch Bly 1.09B 1.30 Bis AG London Barch Bly 1.09B 1.30 Bis AG London Barch Bly 1.09B 1.30 Bis AG London Barch Bly 2.30 1.6 Bis AG London Barch Bly 2.25 2.20 Bis AG London Barch Bly 2.25 2.20 Bis AG London Barch Bly 2.25 2.25 Bis AG London Barch Bly 2.24 2.25 Bis AG London Barch Bly 2.25 2.25<		Buy Buy Buy Buy Buy		629	Ordinary
Bis AG London Branch BLA 6,45 1,000 USS AG London Branch BLA 6,45 1,000 USS AG London Branch BLA 5,135 1,000 USS AG London Branch BLA 1,099 1,099 USS AG London Branch BLA 1,099 1,099 USS AG London Branch BLA 1,099 1,099 USS AG London Branch BLA 1,099 1,09 USS AG London Branch BLA 1,09 1,09 USS AG London Branch BLA 1,09 2,09 USS AG London Branch BLA 2,04 2,09 USS AG London Branch BLA 2,04 2,07 USS AG London Branch BLA 2,04 2,04 USS AG London Branch BLA 2		Buy Buy Buy Buy	5 328		A 1841 144
Bis AG London Branch Buy 5.27 1.02 List AG London Branch Buy 5.135 1.02 List AG London Branch Buy 1.02 1.03 List AG London Branch Buy 1.03 1.03 List AG London Branch Buy 1.03 1.03 List AG London Branch Buy 1.03 1.04 List AG London Branch Buy 1.03 1.04 List AG London Branch Buy 2.74 2.08 List AG London Branch Buy 2.74 2.08 List AG London Branch Buy 2.02 2.03 List AG London Branch Buy 2.04 2.05 List AG London Branch Buy 2.04 2.05 List AG London Branch Buy 2.04 2.04 List AG London Branch Buy 2.04 2.04 List AG London Branch Buy 2.04 2.04 List AG London Branch Buy 2.04 2.05 List AG London Branch Buy		Buy Buy Buy	276,0	000	Ordings
Bis Ad Landon Branch Buy 5,135 677 List Ad Landon Branch Buy 1,159 1,20 List Ad Landon Branch Buy 1,593 1,53 List Ad Landon Branch Buy 1,59 1,59 List Ad Landon Branch Buy 1,59 1,50 List Ad Landon Branch Buy 1,59 1,6 List Ad Landon Branch Buy 1,59 2,2 List Ad Landon Branch Buy 1,5 1,6 List Ad Landon Branch Buy 2,758 3,7 List Ad Landon Branch Buy 2,784 3,7 List Ad Landon Branch Buy 2,785 3,7 List Ad Landon Branch Buy 2,785 3,1 List Ad Landon Branch Buy 2,785 3,2 List Ad Landon Branch Buy <t< td=""><td></td><td>Buy Buy Buy</td><td>000</td><td>100</td><td>Ording.</td></t<>		Buy Buy Buy	000	100	Ording.
URS Act London Branch Bay 1,099 1,09 URS Act London Branch Bay 1,099 1,09 URS Act London Branch Bay 1,099 1,09 URS Act London Branch Bay 1,09 2,09 URS Act London Branch Bay 2,09 3,01 URS Act London Branch Bay 2,09 3,01 URS Act London Branch Bay 2,09 3,01 URS Act London Branch Bay 2,08 3,09 URS Act London Branch Bay 2,08 1,17 URS Act London Branch Bay		Buy	226	200	Olding.
USS ACE LONDON Branch Buy 1,099 USS ACE LONDON Branch Buy 1,099 USS ACE LONDON Branch Buy 1,099 USS ACE LONDON Branch Buy 1,099 USS ACE LONDON Branch Buy 1,099 USS ACE LONDON Branch Buy 2,798 2,784 USS ACE LONDON Branch Buy 2,798 2,784 USS ACE LONDON Branch Buy 2,794 3,78 USS ACE LONDON Branch Buy 2,784 3,07 USS ACE LONDON Branch Buy 2,784 3,07 USS ACE LONDON Branch Buy 2,784 3,07 USS ACE LONDON Branch Buy 2,785 1,17 USS ACE LONDON Branch Buy 2,785 1,17 USS ACE LONDON Branch Buy 2,785 1,10 USS ACE LONDON Branch </td <td></td> <td>buy</td> <td>0010</td> <td>à</td> <td>Cidinaly</td>		buy	0010	à	Cidinaly
USS ACE London Barach Buy 2,391 283 USS ACE London Barach Buy 2,391 2,891 USS ACE London Barach Buy 1,397 2,194 2,981 USS ACE London Barach Buy 1,397 2,194 2,981 USS ACE London Barach Buy 1,394 2,194 3,28 USS ACE London Barach Buy 2,784 3,07 USS ACE London Barach Buy 2,784 3,07 USS ACE London Barach Buy 2,584 3,07 USS ACE London Barach Buy 2,584 3,07 USS ACE London Barach Buy 2,584 3,07 USS ACE London Barach Buy 2,785 3,27 USS ACE London Barach Buy 2,72 3,24			601	2 5	Olderlary
UBS AG London Banch BMY 1,591 2.84 UBS AG London Banch BMY 1,591 2.84 UBS AG London Banch BMY 1,594 2.0 UBS AG London Banch BMY 1,594 2.0 UBS AG London Banch BMY 2,794 3.0 UBS AG London Banch BMY 2,794 3.0 UBS AG London Banch BMY 2,794 4.0 UBS AG London Banch BMY 2,794 4.0 UBS AG London Banch BMY 3,94 4.0 UBS AG London Banch BMY 3,94 2,10 UBS AG London Banch BMY 1,12 1,12 UBS AG London Banch BMY 2,10 2,24 UBS AG London Banch BMY 1,24 1,10 </td <td></td> <td>and.</td> <td>660'I</td> <td>30</td> <td>Ordinary</td>		and.	660'I	30	Ordinary
UBS AG London Branch Bluy 155 25 UBS AG London Branch Bluy 156 27 UBS AG London Branch Bluy 159 20 UBS AG London Branch Bluy 27,88 307 UBS AG London Branch Bluy 27,88 307 UBS AG London Branch Bluy 27,88 307 UBS AG London Branch Bluy 2,584 307 UBS AG London Branch Bluy 2,584 307 UBS AG London Branch Bluy 2,585 45 UBS AG London Branch Bluy 3,584 45 UBS AG London Branch Bluy 3,584 5,042 3,57 UBS AG London Branch Bluy 7,745 3,58 UBS AG London Branch Bluy 7,745 1,68 UBS AG London Branch Bluy 2,166 1,724 UBS AG London Branch Bluy 2,162 1,724 UBS AG London Branch Bluy 2,162 1,724 UBS AG London Branch		Buy	2,391	283	Ordinary
UBS AG London Branch BMy 135 116 UBS AG London Branch BMy 125 15 UBS AG London Branch BMy 160 20 UBS AG London Branch BMy 2,768 328 UBS AG London Branch BMy 2,768 328 UBS AG London Branch BMy 2,768 31 UBS AG London Branch BMy 2,768 328 UBS AG London Branch BMy 2,788 367 UBS AG London Branch BMy 3,288 467 UBS AG London Branch BMy 5,279 3,28 UBS AG London Branch BMy 5,279		Buy	195	23	Ordinary
UBS AG London Banch BM 820 97 UBS AG London Banch Bay 169 20 UBS AG London Banch Bay 177 21 UBS AG London Banch Bay 2,784 30.7 UBS AG London Banch Bay 2,828 46.7 UBS AG London Banch Bay 3,848 46.7 UBS AG London Banch Bay 2,628 3.7 UBS AG London Banch Bay 2,628 3.7 UBS AG London Banch Bay 2,628 3.0 UBS AG London Banch Bay 2,628 3.0 UBS AG London Banch Bay 2,628 3.0 UBS AG London Banch Bay 2,620 1.0 UBS AG London Banch Bay 2,620 1.0 UBS AG London Banch Bay 2,620 1.0		Buy	135	16	Ordinary
UBS AG London Banch Buy 169 20 UBS AG London Banch Buy 734 30 UBS AG London Banch Buy 2,758 328 UBS AG London Banch Buy 2,754 30 UBS AG London Banch Buy 2,598 467 UBS AG London Banch Buy 405 41 UBS AG London Banch Buy 405 42 UBS AG London Banch Buy 5,288 467 UBS AG London Banch Buy 5,288 467 UBS AG London Banch Buy 5,289 47 UBS AG London Banch Buy 5,289 47 UBS AG London Banch Buy 5,289 47 UBS AG London Banch Buy 5,289 5,28 UBS AG London Banch Buy 5,289 5,28 UBS AG London Banch Buy 5,289 1,70 UBS AG London Banch Buy 5,289 1,70 UBS AG London Banch Buy 5,289 1,70 <td></td> <td>Buy</td> <td>820</td> <td>- 64</td> <td>Ordinary</td>		Buy	820	- 64	Ordinary
UBS AG London Banch Bay 794 98 UBS AG London Banch Buy 2,554 307 UBS AG London Banch Buy 2,554 307 UBS AG London Banch Buy 2,554 307 UBS AG London Banch Buy 2,524 31 UBS AG London Banch Buy 95 44 UBS AG London Banch Buy 35,48 467 UBS AG London Banch Buy 35,48 467 UBS AG London Banch Buy 36,49 37,48 467 UBS AG London Banch Buy 7,745 916 92 UBS AG London Banch Buy 2,162 257 113 UBS AG London Banch Buy 2,162 257 145 UBS AG London Banch Buy 2,162 257 145 UBS AG London Banch Buy 2,560 1,010 1,026 UBS AG London Banch Buy 2,560 1,026 1,026 UBS AG London Banch Buy 1,		Buy	1691	20	Ordinary
UBS AG London Branch Boy 2,768 328 UBS AG London Branch Buy 2,594 307 UBS AG London Branch Buy 2,594 307 UBS AG London Branch Buy 408 40 UBS AG London Branch Buy 2,628 31 UBS AG London Branch Buy 3,948 407 UBS AG London Branch Buy 3,948 407 UBS AG London Branch Buy 3,948 5,042 UBS AG London Branch Buy 3,948 5,042 5,97 UBS AG London Branch Buy 3,042 5,042 5,97 UBS AG London Branch Buy 5,042 5,042 5,042 UBS AG London Branch </td <td></td> <td>Ruy</td> <td>797</td> <td>70</td> <td>Ordinary</td>		Ruy	797	70	Ordinary
USS AG London Branch BMY 2.754 3.0 USS AG London Branch BMY 1.77 2.7 USS AG London Branch BMY 1.77 2.7 USS AG London Branch BMY 405 44 USS AG London Branch BMY 3.948 467 USS AG London Branch BW 3.948 3.948 USS AG London Branch BW 3.04 2.042 3.9 USS AG London Branch BW 3.04 2.162 3.9 USS AG London Branch BW 3.04 2.162 3.9 USS AG London Branch BW 3.04 2.533 3.07 USS AG London Branch BW 3.04 1.244 1.710 USS AG London Branch BW 3.54 4.04 1.724 USS AG London Branch BW 3.533		Pars.	27.6	330	Ording
USS AG London Branch Buy 17.74 21.1 USS AG London Branch Buy 1.25 21.1 USS AG London Branch Buy 405 2.62 1.1 USS AG London Branch Buy 2.62 1.1 USS AG London Branch Buy 2.62 9.2 USS AG London Branch Buy 2.62 9.2 USS AG London Branch Buy 2.74S 1.75 USS AG London Branch Buy 2.74S 1.75 USS AG London Branch Buy 1.02B 1.71B USS AG London Branch Buy 1.02B 1.71B USS AG London Branch Buy 1.20G 1.71 USS AG London Branch Buy 1.20G 1.72 USS AG London Branch Buy <td></td> <td>200</td> <td>2,708</td> <td>200</td> <td>Ciderialy</td>		200	2,708	200	Ciderialy
URS AG London Branch Buy 177 171 URS AG London Branch Buy 93 14 URS AG London Branch Buy 405 44 URS AG London Branch Buy 262 31 URS AG London Branch Buy 262 131 URS AG London Branch Buy 262 131 URS AG London Branch Buy 285 925 URS AG London Branch Buy 2165 132 URS AG London Branch Buy 2162 256 URS AG London Branch Buy 2162 256 URS AG London Branch Buy 2162 257 URS AG London Branch Buy 2166 1174 URS AG London Branch Buy 2168 1276 URS AG London Branch Buy 2168 128 URS AG London Branch Buy 2168 128 URS AG London Branch Buy 2168 128 URS AG London Branch Buy 2160 118 </td <td>111</td> <td>Buy</td> <td>2,594</td> <td>30/</td> <td>Ordinary</td>	111	Buy	2,594	30/	Ordinary
URS AG London Branch Buy 93 41 URS AG London Branch Buy 405 41 URS AG London Branch Buy 262 41 URS AG London Branch Buy 262 923 URS AG London Branch Buy 3.948 467 URS AG London Branch Buy 7.745 916 URS AG London Branch Buy 7.745 916 URS AG London Branch Buy 2.162 259 URS AG London Branch Buy 2.162 1.706 URS AG London Branch Buy 2.102 1.701 URS AG London Branch Buy 1.704 1.701 URS AG London Branch Buy 1.706 1.706 URS AG London Branch Buy 2.506 1.707 URS AG London Branch Buy 1.700		euy	177	21	Ordinary
UBS AG London Branch Buy 405 418 UBS AG London Branch Buy 405 411 UBS AG London Branch Buy 202 31 UBS AG London Branch Buy 2848 467 UBS AG London Branch Buy 7,745 916 UBS AG London Branch Buy 2,166 1,724 UBS AG London Branch Buy 2,168 1,028 UBS AG London Branch Buy 2,169 1,734 UBS AG London Branch Buy 8,504 1,010 UBS AG London Branch Buy 8,504 1,010 UBS AG London Branch Buy 8,504 1,010 UBS AG London Branch Buy 4,507 2,509 1,010 UBS AG London Branch Buy<	\top	Buy	603	=	Ordinary
UBS ACL tondon Branch Buy 263 11 UBS ACL tondon Branch Buy 263 11 UBS ACL tondon Branch Buy 3,948 467 UBS ACL tondon Branch Buy 3,948 467 UBS ACL tondon Branch Buy 7,745 929 UBS ACL tondon Branch Buy 7,745 916 UBS ACL tondon Branch Buy 7,745 916 UBS ACL tondon Branch Buy 2,742 3,94 UBS ACL tondon Branch Buy 2,162 2,54 UBS ACL tondon Branch Buy 2,162 2,54 UBS ACL tondon Branch Buy 2,162 1,74 1,71 UBS ACL tondon Branch Buy 6,04 2,504 1,72 UBS ACL tondon Branch Buy 8,24 1,71 1,71 UBS ACL tondon Branch Buy 1,691 1,71 UBS ACL tondon Branch Buy 2,593 3,74 4,28 UBS ACL tondon Branch Buy 1,691 1,744<		Buy	405	48	Ordinary
UBS AG London Branch Buy 262 31 UBS AG London Branch Buy 3,948 467 UBS AG London Branch Buy 7,855 67 UBS AG London Branch Buy 7,751 6 UBS AG London Branch Buy 7,745 916 UBS AG London Branch Buy 2,166 77 UBS AG London Branch Buy 2,162 254 UBS AG London Branch Buy 6,042 5,042 5,04 UBS AG London Branch Buy 1,174 1,18 UBS AG London Branch Buy 1,206 1,19 UBS AG London Branch Buy 1,206 1,19 UBS AG London Branch Buy 1,206 1,19 UBS AG London Branch Buy 1,206 1,206 UBS AG London Branch Buy		Buy	66	11	Ordinary
URS AC London Branch Buy 3,348 467 UBS AC London Branch Buy 7,885 929 UBS AC London Branch Buy 7,885 929 UBS AC London Branch Buy 7,745 916 UBS AC London Branch Buy 7,745 916 UBS AC London Branch Buy 7,745 917 UBS AC London Branch Buy 2,162 297 UBS AC London Branch Buy 2,162 297 UBS AC London Branch Buy 2,162 256 UBS AC London Branch Buy 9,256 1,036 UBS AC London Branch Buy 9,276 1,131 UBS AC London Branch Buy 1,204 1,204 UBS AC London Branch Buy 1,206 1,939 UBS AC London Branch Buy 2,593 307 UBS AC London Branch Buy 2,593 307 UBS AC London Branch Buy 4,593 2,504 1,036 UBS AC London Branch		Buy	262	31	Ordinary
UBS ACI Lendon Branch Buy 7,855 929 UBS ACI Lendon Branch Buy 7,855 929 UBS ACI Lendon Branch Buy 5745 916 UBS ACI Lendon Branch Buy 7,745 916 UBS ACI Lendon Branch Buy 2,762 257 UBS ACI Lendon Branch Buy 2,166 275 UBS ACI Lendon Branch Buy 2,166 173 UBS ACI Lendon Branch Buy 2,166 173 UBS ACI Lendon Branch Buy 6,08 1,174 1,37 UBS ACI Lendon Branch Buy 1,174 1,39 UBS ACI Lendon Branch Buy 1,206 1,174 UBS ACI Lendon Branch Buy 2,259 1,174 UBS ACI Lendon Branch Buy 2,250 1,274 UBS ACI Lendon Branc	Г	Buv	3 948	467	Ordinary
UBS AG London Branch Buy 955 173 UBS AG London Branch Buy 955 173 UBS AG London Branch Buy 7745 916 UBS AG London Branch Buy 7,745 916 UBS AG London Branch Buy 2,162 256 UBS AG London Branch Buy 2,162 256 UBS AG London Branch Buy 6,09 7,745 916 UBS AG London Branch Buy 6,08 7,72 1,09 UBS AG London Branch Buy 6,08 7,72 UBS AG London Branch Buy 1,174 1,70 UBS AG London Branch Buy 1,441 1,70 UBS AG London Branch Buy 1,444 1,70 UBS AG London Branch Buy 1,89 1,89 UBS AG London Branch Buy 2,59 1,81 UBS AG London Branch Buy 4,44B 5,27 UBS AG London Branch Buy 4,591 1,724 UBS AG London Branch	T	N. W.	7 955	020	Ording.
UBS AG London Branch Buy 51 16 UBS AG London Branch Buy 7745 91 UBS AG London Branch Buy 7745 91 UBS AG London Branch Buy 278 33 UBS AG London Branch Buy 278 33 UBS AG London Branch Buy 2.162 2.76 UBS AG London Branch Buy 2.162 2.75 UBS AG London Branch Buy 1.714 1.99 UBS AG London Branch Buy 1.441 1.710 UBS AG London Branch Buy 1.2056 1.191 UBS AG London Branch Buy 1.2056 1.191 UBS AG London Branch Buy 1.2056 1.191 UBS AG London Branch Buy 1.591 1.91 UBS AG London Branch Buy 1.591 1.81 UBS AG London Branch Buy 2.593 3.74 UBS AG London Branch Buy 2.594 4.436 UBS AG London Branch Buy 2.597 </td <td>Ť</td> <td>, , ,</td> <td>330</td> <td>54,</td> <td>O C</td>	Ť	, , ,	330	54,	O C
UBS AG London Branch Buy 7745 9 b UBS AG London Branch Buy 18 303 2,166 UBS AG London Branch Buy 18 303 2,166 UBS AG London Branch Buy 5,042 559 UBS AG London Branch Buy 5,042 559 UBS AG London Branch Buy 5,045 1,70 UBS AG London Branch Buy 6,04 6,04 1,71 UBS AG London Branch Buy 1,441 1,710 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 1,398 1,339 UBS AG London Branch Buy 1,398 1,339 UBS AG London Branch Buy 4,448 5,27 UBS AG London Branch Buy 4,597 2,644 UBS AG London Branch Buy 4,448 5,27 UBS AG London Branch	T	600	500	2 4	Cidillary
UBS AG London Branch Buy 1745 916 UBS AG London Branch Buy 1745 916 UBS AG London Branch Buy 279 237 UBS AG London Branch Buy 2.162 2.97 UBS AG London Branch Buy 2.162 2.95 UBS AG London Branch Buy 2.162 1.75 UBS AG London Branch Buy 9.256 1.058 UBS AG London Branch Buy 9.256 1.026 UBS AG London Branch Buy 1.744 1.724 UBS AG London Branch Buy 1.2026 1.72 UBS AG London Branch Buy 2.593 3.07 UBS AG London Branch Buy 2.593 3.04 UBS AG London Branch Buy 2.593 3.04 UBS AG London Branch Buy 4.753 4.85 UBS AG London Branch Buy 4.753 2.84 UBS AG London Branch Buy 4.753 2.050 UBS AG London Branch Buy <t< td=""><td>T</td><td>, and</td><td>ic</td><td>٥</td><td>Ordinary</td></t<>	T	, and	ic	٥	Ordinary
UBS AG London Branch Buy 189 189 2166 UBS AG London Branch Buy 279 233 UBS AG London Branch Buy 276 259 236 UBS AG London Branch Buy 2,162 256 1056 UBS AG London Branch Buy 650 77 UBS AG London Branch Buy 608 1,75 UBS AG London Branch Buy 1,174 1,39 UBS AG London Branch Buy 1,628 1,006 UBS AG London Branch Buy 6,08 1,010 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 1,006 1,19 UBS AG London Branch Buy 1,39 1,339 UBS AG London Branch Buy 1,39 1,339 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,597 5,05 UBS AG London Branch Buy 4,048 5,27 UBS	- 1	Buy	7,745	916	Ordinary
UBS AG London Branch Buy 5.79 33 UBS AG London Branch Buy 5.042 5.97 UBS AG London Branch Buy 2.162 2.56 UBS AG London Branch Buy 6.08 7.72 UBS AG London Branch Buy 6.08 7.72 UBS AG London Branch Buy 6.08 7.72 UBS AG London Branch Buy 1.174 1.70 UBS AG London Branch Buy 1.62.05 1.424 UBS AG London Branch Buy 1.62.05 1.74 UBS AG London Branch Buy 8.540 1.010 UBS AG London Branch Buy 2.593 3.07 UBS AG London Branch Buy 2.593 3.07 UBS AG London Branch Buy 4.591 5.24 UBS AG London Branch Buy 4.591 5.24 UBS AG London Branch Buy 4.591 5.24 UBS AG London Branch Buy 4.591 5.27 UBS AG London Branch Buy	- 1	Buy	18,303	2,166	Ordinary
UBS AG London Branch Buy 5,042 597 UBS AG London Branch Buy 2,162 256 UBS AG London Branch Buy 2,162 2,76 UBS AG London Branch Buy 1,174 1,99 UBS AG London Branch Buy 1,441 1,710 UBS AG London Branch Buy 1,2056 1,710 UBS AG London Branch Buy 8,581 1,010 UBS AG London Branch Buy 8,581 1,010 UBS AG London Branch Buy 8,581 1,010 UBS AG London Branch Buy 5,593 307 UBS AG London Branch Buy 1,891 1,891 2,24 UBS AG London Branch Buy 1,891 1,891 2,24 UBS AG London Branch Buy 4,897 4,897 5,84 UBS AG London Branch Buy 4,937 4,82 UBS AG London Branch Buy 4,073 4,82 UBS AG London Branch Buy 3,27 1,741		Buy	279	33	Ordinary
UBS AG London Branch Buy 2.162 2.56 UBS AG London Branch Buy 6.50 7.73 UBS AG London Branch Buy 6.50 7.73 UBS AG London Branch Buy 6.08 7.22 UBS AG London Branch Buy 6.08 7.22 UBS AG London Branch Buy 6.08 7.22 UBS AG London Branch Buy 8.540 1.010 UBS AG London Branch Buy 8.540 1.010 UBS AG London Branch Buy 8.540 1.010 UBS AG London Branch Buy 1.891 2.24 UBS AG London Branch Buy 1.308 1.339 UBS AG London Branch Buy 4.597 1.448 5.24 UBS AG London Branch Buy 4.597 4.073 5.64 UBS AG London Branch Buy 4.73 4.82 1.08 UBS AG London Branch Buy 4.448 5.27 4.08 1.08 UBS AG London Branch Buy 4.448 <td></td> <td>9nv</td> <td>5.042</td> <td>265</td> <td>Ordinary</td>		9nv	5.042	265	Ordinary
UBS AG London Branch Buy 2.162 2.56 UBS AG London Branch Buy 2.162 2.56 UBS AG London Branch Buy 9,256 1,096 UBS AG London Branch Buy 1,441 1,710 UBS AG London Branch Buy 14,441 1,710 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 1,006 1,19 UBS AG London Branch Buy 1,691 1,91 UBS AG London Branch Buy 1,691 1,24 UBS AG London Branch Buy 1,691 1,24 UBS AG London Branch Buy 1,691 1,39 UBS AG London Branch Buy 1,691 1,39 UBS AG London Branch Buy 4,597 2,51 UBS AG London Branch Buy 4,445 5,27 UBS AG London Branch Buy 2,374 2,81 UBS AG London Branch Buy	L	2 4	35	0	Ordingo
UBS AG London Branch Buy 6.16.2 2.50 UBS AG London Branch Buy 6.50 7.70 UBS AG London Branch Buy 6.08 7.2 UBS AG London Branch Buy 6.08 7.2 UBS AG London Branch Buy 6.08 7.2 UBS AG London Branch Buy 8.681 1.010 UBS AG London Branch Buy 8.540 1.010 UBS AG London Branch Buy 8.540 1.010 UBS AG London Branch Buy 1.00 1.9 UBS AG London Branch Buy 1.530 1.339 UBS AG London Branch Buy 4.597 1.339 UBS AG London Branch Buy 4.597 6.44 UBS AG London Branch Buy 4.733 5.64 UBS AG London Branch Buy 4.073 4.62 UBS AG London Branch Buy 4.711 1.744 UBS AG London Branch Buy 4.448 5.27 UBS AG London Branch Buy <t< td=""><td>T</td><td>600</td><td>67</td><td>0 22</td><td>000</td></t<>	T	600	67	0 22	000
UBS AG London Branch Buy 9.256 1.097 UBS AG London Branch Buy 9.256 1.096 UBS AG London Branch Buy 1.174 1.39 UBS AG London Branch Buy 16.026 1.424 UBS AG London Branch Buy 12.026 1.424 UBS AG London Branch Buy 8.581 1.018 UBS AG London Branch Buy 8.599 1.181 UBS AG London Branch Buy 2.593 3.07 UBS AG London Branch Buy 1,891 2.54 UBS AG London Branch Buy 4,591 5.24 UBS AG London Branch Buy 4,733 4.85 UBS AG London Branch Buy 4,733 4.85 UBS AG London Branch Buy 4,733 4.85 UBS AG London Branch Buy 4,448 5.71 UBS AG London Branch Buy 4,448 5.74 UBS AG London Branch Buy 4,448 5.74 UBS AG London Branch Buy	Ť	on's	7,192	007	Orginary
UBS AG London Branch Buy 9,256 1,096 UBS AG London Branch Buy 1,174 1,39 UBS AG London Branch Buy 1,441 1,710 UBS AG London Branch Buy 1,441 1,710 UBS AG London Branch Buy 8,581 1,008 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 1,006 1,19 UBS AG London Branch Buy 2,593 307 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,733 5,6 UBS AG London Branch Buy 4,733 5,6 UBS AG London Branch Buy 4,448 5,27 UBS AG London Branch Buy 4,448 5,27 UBS AG London Branch Buy 4,448 5,27 UBS AG London Branch Buy 2,592 2,590 UBS AG London Branch Buy	7	Buy	650	77	Ordinary
UBS AG London Branch Buy 1.74 139 UBS AG London Branch Buy 608 7.2 UBS AG London Branch Buy 1.4441 1.710 UBS AG London Branch Buy 8.581 1.010 UBS AG London Branch Buy 8.540 1.010 UBS AG London Branch Buy 8.5979 1.181 UBS AG London Branch Buy 2.593 307 UBS AG London Branch Buy 1.891 2.44 UBS AG London Branch Buy 4.591 5.44 UBS AG London Branch Buy 4.591 5.44 UBS AG London Branch Buy 4.591 5.24 UBS AG London Branch Buy 4.733 482 UBS AG London Branch Buy 4.748 5.27 UBS AG London Branch Buy <	7	Buy	9,256	1,096	Ordinary
UBS AG London Branch Buy 14,441 1,710 UBS AG London Branch Buy 14,441 1,710 UBS AG London Branch Buy 8,540 1,012 UBS AG London Branch Buy 9,72 1,101 UBS AG London Branch Buy 9,72 1,119 UBS AG London Branch Buy 1,006 1,19 UBS AG London Branch Buy 1,891 2,24 UBS AG London Branch Buy 1,891 2,24 UBS AG London Branch Buy 1,891 2,24 UBS AG London Branch Buy 4,597 54 UBS AG London Branch Buy 4,733 56 UBS AG London Branch Buy 4,733 56 UBS AG London Branch Buy 4,733 57 UBS AG London Branch Buy 5,374 281 UBS AG London Branch Buy 3,27 2,050 UBS AG London Branch Buy 3,27 2,050 UBS AG London Branch Buy	┪	Buy	1,174	139	Ordinary
UBS AG London Branch Buy 14,41 1,710 UBS AG London Branch Buy 16,205 1,424 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 2,593 307 UBS AG London Branch Buy 1,891 25,91 UBS AG London Branch Buy 1,891 2,591 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,073 4,82 UBS AG London Branch Buy 4,073 4,82 UBS AG London Branch Buy 4,073 4,82 UBS AG London Branch Buy 4,448 5,27 UBS AG London Branch Buy		Buv	809	72	Ordinary
UBS AG London Branch Buy 1,2,226 1,424 UBS AG London Branch Buy 8,681 1,018 UBS AG London Branch Buy 8,540 1,101 UBS AG London Branch Buy 9,979 1,181 UBS AG London Branch Buy 1,591 2,24 UBS AG London Branch Buy 1,597 5,44 UBS AG London Branch Buy 4,73 42 UBS AG London Branch Buy 5,40 64 UBS AG London Branch Buy 5,44 5,7 UBS AG London Branch Buy 5,27 2,050 UBS AG London Branch Buy 5,27 2,050 UBS AG London Branch Buy 5,27 2,050 UBS AG London Branch Buy 5,27 1,056 UBS AG London Branch Buy 5,27 <td>Ħ</td> <td>N. B.</td> <td>14 441</td> <td>1 210</td> <td>Ordinan</td>	Ħ	N. B.	14 441	1 210	Ordinan
UBS AG London Branch Buy 8,2425 1,424 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 9,979 1,181 UBS AG London Branch Buy 2,593 307 UBS AG London Branch Buy 1,891 2,593 UBS AG London Branch Buy 1,392 1,339 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,073 5,6 UBS AG London Branch Buy 4,448 5,27 UBS AG London Branch Buy 4,448 5,274 UBS AG London Branch Buy	T	â	144,41	0.7,	Coll lay
UBS AG London Branch Buy 8,681 1,028 UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 1,006 119 UBS AG London Branch Buy 2,593 307 UBS AG London Branch Buy 1,891 2.24 UBS AG London Branch Buy 4,397 5,54 UBS AG London Branch Buy 4,73 482 UBS AG London Branch Buy 4,73 482 UBS AG London Branch Buy 4,73 56 UBS AG London Branch Buy 4,48 5.7 UBS AG London Branch Buy 4,48 5.7 UBS AG London Branch Buy 3,29 3,9 UBS AG London Branch Buy 3,29 3,0 UBS AG London Branch Buy 3,29 3,0 UBS AG London Branch Buy 3,23 3,0 UBS AG London Branch Buy 3,23 3,0 UBS AG London Branch Buy 3,23	1	euy	12,026	474	Ordinary
UBS AG London Branch Buy 8,540 1,010 UBS AG London Branch Buy 9,979 1,181 UBS AG London Branch Buy 2,593 307 UBS AG London Branch Buy 1,891 2,24 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,073 4,82 UBS AG London Branch Buy 4,073 4,82 UBS AG London Branch Buy 4,448 5,27 UBS AG London Branch Buy 5,374 2,81 UBS AG London Branch Buy 3,374 2,81 UBS AG London Branch Buy 3,374 2,81 UBS AG London Branch Buy 1,7292 2,050 UBS AG London Branch Buy 1,7292 2,050 UBS AG London Branch Buy 1,7292 2,050 UBS AG London Branch Buy 5,374 36 UBS AG London Branch Buy		Buy	8,681	1,028	Ordinary
UBS AG London Branch Buy 9.979 1,181 UBS AG London Branch Buy 1,006 119 UBS AG London Branch Buy 1,891 2,593 UBS AG London Branch Buy 1,891 2,54 UBS AG London Branch Buy 4,597 544 UBS AG London Branch Buy 4,73 56 UBS AG London Branch Buy 4,073 43 UBS AG London Branch Buy 7,6 9 UBS AG London Branch Buy 4,44B 57 UBS AG London Branch Buy 1,732 2,05 UBS AG London Branch Buy 5,34 2,05 UBS AG London Branch Buy 5,226 7,03 UBS AG London Branch Buy 5,326		Buy	8,540	1,010	Ordinary
UBS AG Landon Branch Buy 1,006 1,19 UBS AG Landon Branch Buy 1,593 307 UBS AG London Branch Buy 1,691 224 UBS AG London Branch Buy 1,891 2,24 UBS AG London Branch Buy 4,597 54 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 5,374 281 UBS AG London Branch Buy 2,374 281 UBS AG London Branch Buy 3,274 3,65 UBS AG London Branch Buy 3,274 3,65 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,226		Riv	979.0	1 181	Ordinary
UBS AG London Branch Buy 1,000 UBS AG London Branch Buy 2,593 317 UBS AG London Branch Buy 1,891 2,544 UBS AG London Branch Buy 4,597 5,44 UBS AG London Branch Buy 4,737 56 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 5,40 9 UBS AG London Branch Buy 4,043 57 UBS AG London Branch Buy 4,448 57 UBS AG London Branch Buy 4,44B 57 UBS AG London Branch Buy 4,44B 57 UBS AG London Branch Buy 4,44B 57 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,226 7,03 </td <td>1</td> <td></td> <td>500 1</td> <td>5.</td> <td>(a)</td>	1		500 1	5.	(a)
UBS AG London Branch Buy 1,593 307 UBS AG London Branch Buy 1,691 124 UBS AG London Branch Buy 1,891 224 UBS AG London Branch Buy 4,73 482 UBS AG London Branch Buy 4,73 482 UBS AG London Branch Buy 7,6 9 UBS AG London Branch Buy 4,44 5,7 UBS AG London Branch Buy 4,44 5,7 UBS AG London Branch Buy 4,44 5,7 UBS AG London Branch Buy 2,374 339 UBS AG London Branch Buy 3,29 305 UBS AG London Branch Buy 3,74 32 UBS AG London Branch Buy 3,7 1,056 UBS AG London Branch Buy 5,226	Ť	ânâ	900,1	6 :	Orginary
UBS AG London Branch Buy 160 19 UBS AG London Branch Buy 1,891 224 UBS AG London Branch Buy 4,597 544 UBS AG London Branch Buy 4,737 482 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 4,448 527 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 1,711 1,744 UBS AG London Branch Buy 5,226 1,056 UBS AG London Branch Buy 5,226 1,056 UBS AG London Branch Buy <	1	Buy	2,593	307	Ordinary
UBS AG London Branch Buy 1,891 224 UBS AG London Branch Buy 4,597 544 UBS AG London Branch Buy 4,73 58 UBS AG London Branch Buy 4,73 482 UBS AG London Branch Buy 7,6 64 UBS AG London Branch Buy 4,48 527 UBS AG London Branch Buy 2,374 32 UBS AG London Branch Buy 3,29 39 UBS AG London Branch Buy 1,744 20 UBS AG London Branch Buy 3,29 30 UBS AG London Branch Buy 3,29 30 UBS AG London Branch Buy 3,29 30 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,226 <t< td=""><td></td><td>Buy</td><td>160</td><td>19</td><td>Ordinary</td></t<>		Buy	160	19	Ordinary
UBS AG London Branch Buy 4,597 544 UBS AG London Branch Buy 1,308 1,339 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 540 64 UBS AG London Branch Buy 540 64 UBS AG London Branch Buy 4,448 527 UBS AG London Branch Buy 3,274 281 UBS AG London Branch Buy 1,7,292 2,050 UBS AG London Branch Buy 1,7,292 2,050 UBS AG London Branch Buy 1,7,292 2,050 UBS AG London Branch Buy 5,226 1,056 UBS AG London Branch Buy 8,239 39 UBS AG London Branch Buy <	Г	Silv	1 891	727	Ordinary
UBS AG London Branch Buy 1,329 1,339 UBS AG London Branch Buy 473 56 UBS AG London Branch Buy 76 9 UBS AG London Branch Buy 540 64 UBS AG London Branch Buy 2,374 57 UBS AG London Branch Buy 2,374 51 UBS AG London Branch Buy 2,374 32 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 329 58 UBS AG London Branch Buy 329 1,096 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 2,393 394 UBS AG London Branch Buy 2,393 <	T	2010	A E07	777	i i i
UBS AG London Branch Buy 1,308 1,339 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 540 64 UBS AG London Branch Buy 4,448 527 UBS AG London Branch Buy 3,274 281 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 5,286 65 UBS AG London Branch Buy 5,226 703 UBS AG London Branch Buy 2,393 <td>T</td> <td>, and a</td> <td>160,4</td> <td></td> <td>O CILITAL A</td>	T	, and a	160,4		O CILITAL A
UBS AG London Branch Buy 473 56 UBS AG London Branch Buy 4,073 482 UBS AG London Branch Buy 76 9 UBS AG London Branch Buy 540 64 UBS AG London Branch Buy 2,374 281 UBS AG London Branch Buy 2,324 281 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 5,266 1,066 UBS AG London Branch Buy 5,228 1,066 UBS AG London Branch Buy 5,228 1,066 UBS AG London Branch Buy 5,226 7,03 UBS AG London Branch Buy 5,226 7,09 UBS AG London Branch Buy 5,226 7,09 UBS AG London Branch Buy 5,226 7,09 UBS AG London Branch Buy 5,226 <th< td=""><td>1</td><td>Buy</td><td>11,308</td><td>1,339</td><td>Ordinary</td></th<>	1	Buy	11,308	1,339	Ordinary
UBS AG London Branch Buy 4,073 482 USS AG London Branch Buy 76 64 UBS AG London Branch Buy 540 64 UBS AG London Branch Buy 4,445 527 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 329 65 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,926 7,03 UBS AG London Branch Buy 8,228 1,096 UBS AG London Branch Buy 8,23 1,00 UBS AG London Branch Buy 8,23 1,00 UBS AG London Branch Buy 2,393 364 UBS AG London Branch Buy 3,597	٦	Buy	473	95	Ordinary
UBS AG London Branch Buy 76 9 UBS AG London Branch Buy 540 64 UBS AG London Branch Buy 4,448 527 UBS AG London Branch Buy 329 281 UBS AG London Branch Buy 17,232 2,050 UBS AG London Branch Buy 17,732 2,050 UBS AG London Branch Buy 542 65 UBS AG London Branch Buy 542 65 UBS AG London Branch Buy 5,226 7,03 UBS AG London Branch Buy 8,236 7,03 UBS AG London Branch Buy 8,226 7,03 UBS AG London Branch Buy 8,32 1,096 UBS AG London Branch Buy 8,32 1,00 UBS AG London Branch Buy 2,393 394 UBS AG London Branch Buy 2,393 364	_	Buy	4,073	482	Ordinary
UBS AG London Branch Buy 540 64 UBS AG London Branch Buy 448 527 UBS AG London Branch Buy 2,374 281 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 5,228 1,096 UBS AG London Branch Buy 5,228 1,096 UBS AG London Branch Buy 5,228 1,096 UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 2,393 394 UBS AG London Branch Buy 2,393 394 UBS AG London Branch Buy 3,64 3,64	Ī	Rus	75	0	Ordinary
UBS AG London Branch Buy 4,448 571 UBS AG London Branch Buy 2,374 281 UBS AG London Branch Buy 2,374 281 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 17,732 2,050 UBS AG London Branch Buy 14,711 1,744 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 5,226 7,03 UBS AG London Branch Buy 8,32 1,096 UBS AG London Branch Buy 8,32 1,096 UBS AG London Branch Buy 8,32 1,09 UBS AG London Branch Buy 2,393 39 UBS AG London Branch Buy 2,393 364 UBS AG London Branch Buy 2,393 364	t	, , ,	240		(in the contract of
UBS AG London Branch Buy 2,34 281 UBS AG London Branch Buy 2,34 281 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 5,48 65 UBS AG London Branch Buy 5,226 7,09 UBS AG London Branch Buy 5,926 7,09 UBS AG London Branch Buy 2,393 394 UBS AG London Branch Buy 2,393 364 UBS AG London Branch Buy 3,667 364	Ť	had	240	40	Ordinary
UBS AG London Branch Buy 2374 281 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 14,711 1,744 UBS AG London Branch Buy 5,29 39 UBS AG London Branch Buy 5,28 1,096 UBS AG London Branch Buy 5,226 1,096 UBS AG London Branch Buy 843 1,00 UBS AG London Branch Buy 843 1,00 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 3,29 39 UBS AG London Branch Buy 3,59 364	7	buy	4,448	527	Ordinary
UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 14,711 1,744 UBS AG London Branch Buy 548 65 UBS AG London Branch Buy 5,226 7,035 UBS AG London Branch Buy 5,926 7,035 UBS AG London Branch Buy 5,926 7,035 UBS AG London Branch Buy 2,393 394 UBS AG London Branch Buy 2,393 364 UBS AG London Branch Buy 3,647 364	=	Buy	2,374	781	Ordinary
UBS AG London Branch Buy 17,292 2,050 UBS AG London Branch Buy 1,741 1,744 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 548 65 UBS AG London Branch Buy 5,228 1,096 UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 843 100 UBS AG London Branch Buy 8,39 39 UBS AG London Branch Buy 2,393 384 UBS AG London Branch Buy 3,667 364	_	Buy	329	39	Ordinary
UBS AG London Branch Buy 14,711 1,744 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 548 65 UBS AG London Branch Buy 9,228 1,096 UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 843 100 UBS AG London Branch Buy 2,393 39 UBS AG London Branch Buy 2,393 364 UBS AG London Branch Buy 3,677 364	┢	Rin	17 292	2.050	Ordinary
UBS AG London Branch Buy 3.24 UBS AG London Branch Buy 3.28 65 UBS AG London Branch Buy 5.228 1.096 UBS AG London Branch Buy 5.226 1.096 UBS AG London Branch Buy 5.926 1.006 UBS AG London Branch Buy 8.43 1.00 UBS AG London Branch Buy 8.29 39 UBS AG London Branch Buy 2.393 284 UBS AG London Branch Buy 3.067 364			17,224	2,000	in in
UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 548 106 UBS AG London Branch Buy 9,228 1,096 UBS AG London Branch Buy 5,926 7,03 UBS AG London Branch Buy 843 100 UBS AG London Branch Buy 2,393 294 UBS AG London Branch Buy 3,657 364	-	gris	14,/11	1,/44	Ordinary
UBS AG London Branch Buy 548 65 UBS AG London Branch Buy 9.228 1,096 UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 843 100 UBS AG London Branch Buy 2,393 39 UBS AG London Branch Buy 2,393 244 UBS AG London Branch Buy 3,64 364	١	Buy	329	39	Ordinary
UBS AG London Branch Buy 9,228 1,096 UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 843 100 UBS AG London Branch Buy 2,339 39 UBS AG London Branch Buy 3,667 364		Buy	248	9	Ordinary
UBS AG London Branch Buy 5,926 703 UBS AG London Branch Buy 843 100 UBS AG London Branch Buy 2,393 39 UBS AG London Branch Buy 2,393 244 UBS AG London Branch Buy 3,667 364	Г	Ang Ang	9.228	1.096	Ordinary
UBS AG London Branch Buy 843 100 UBS AG London Branch Buy 329 39 UBS AG London Branch Buy 2,393 284 UBS AG London Branch Buy 3,667 364	Г	812	5,926	703	Ordinary
UBS AG London Branch Buy 329 100 UBS AG London Branch Buy 3.29 284 UBS AG London Branch Buy 2.393 284 UBS AG London Branch Buy 3.667 364	П		0.2010	300	i i
UBS AG London Branch Buy 2,393 39 USS AG London Branch Buy 2,393 284 UBS AG London Branch Buy 3,067 364	1	any	843	3	Ordinary
UBS AG London Branch Buy 2,393 284 UBS AG London Branch Buy 3,067 364	- 1	Buy	329	39	Ordinary
UBS AG London Branch Buy 3,067 364		Buy	2,393	284	Ordinary
	Г	71.0	3.067	36.4	Ordina
	T	auy	/an/c	906	Ordinary

UBS AG London Branch		1,197	Ordinary
	Buy 2,819	332	Ordinary
		971	Ordinary
	8uv 6 708	791	Ordinary
		523	Ordinary
	Buy 2,190	258	Ordinary
		1,454	Ordinary
	Buy 1,424	168	Ordinary
		7	Ordinary
	5/3 5	44	Ordinary
			Ordinary
	ıci	791	Ordinary
		12	Ordinary
		160	Ordinary
		218	Ordinany
	1717 S 2 217	Vac	Ordinan
		33	Ordinan
l		111	1000
l		1	Cidillary
l		۵	Orginary
		1,165	Ordinary
	Buy 4,766	263	Ordinary
		761	Ordinary
		136	Ordinan
l		171	Ordinan
١		-	Coding
1			Ordinary
İ		/3	Ordinary
		135	Ordinary
	Buy 1,216	144	Ordinary
		465	Ordinary
		å	Ordinary
		5	Ordinary
		3 5	in in
1	Buy 1,366	162	Ordinary
		244	Ordinary
	Buy 835	66	Ordinary
		128	Ordinary
	787	22	Ordinan
		2	Oldinary
		/6	Ordinary
	1,544	183	Ordinary
		104	Ordinary
		13	Ordinary
			Carling
			Collinary
	120,1	171	Ordinary
	-	8	Ordinary
		69	Ordinary
ļ		96	Cardinago
	020	2	Cidirial y
		37	Ordinary
	Buy 17	2	Ordinary
	Buy 261	31	Ordinary
		٥	Ordinan
	(2)	•	Countries y
İ			Ordinary
	Buy 1,234	146	Ordinary
	8uy 600	71	Ordinary
	1.073	127	Ordinary
		3	, ,
		60	Ordinary
	1,420	168	Ordinary
	Buy 2,983	323	Ordinary
		295	Ordinary
l		3.0	Ordinary
		7	Ordinally
	Buy 913	108	Ordinary
		142	Ordinary
	Buy 964	114	Ordinary
		30	Ordinary
		22.0	
	3,1//	376	Ordinary
	3uy 811	96	Ordinary

04-Apr-14	I IDC Constition A settential ted	C	****		
24-Apr-14	UBS Securities Australia Itd	Enter Into Derivative	VIN	4 6	Coptions
04-Apr-14	UBS Securities Australia Ltd	Sell	47,024	(5,565)	Ordinary
04-Apr-14	UBS Securities Australia Ltd	Buy	101	12	Ordinary
04-Apr-14	UBS Securities Australia Ltd	Buv	1.859	220	Ordinary
34-Apr-14	UBS Securities Australia Ltd	Sell	84,880	(10,045)	Ordinary
34-Apr-14	UBS Securities Australia Ltd	Sell	28.164	(3.333)	Ordinary
-Apr-14	UBS AG Australia Branch	Sell	2.311	(274)	Ordinary
7-Apr-14	UBS AG Australia Branch	Sell	33,153	(13,921)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	10.920	(1,300)	Ordinary
37-Apr-14	UBS Securities Australia Ltd	Buv	79 565	3 507	Ordinary
-Apr-14	UBS Securities Australia Ltd	Buv	395,750	46.897	Ordinary
-Apr-14	UBS Securities Australia Ltd	13	448 528	(53.052)	Ordinary
07-Apr-14	UBS Securities Australia Ltd	Buv	31.056	3 700	Grain
7-Apr-14	UBS Securities Australia Ltd	Ruy	151 535	300 CF	Ordinary
-Apr-14	LIBS Securities Australia Ltd	Coll	21 150	42,360	Ordinary
-Apr-14	LIRS Securities Australia 11d	Coll	100 400	(2000)	Grainary
72-Apr-14	LIRC Countries Australia Ltd	Cater late Demostra	109,423	(22,450)	Ordinary
7-Apr-14	I IRC Counties Australia 1-4	Enter Into Derivative	(A)	(1,100)	Options
1 7	USO Countries Australia Ltd	Cater into Delivative of	YA!	001	Options
7-Apr-14	1 195 Societies Australia 14d	Enter into Derivative	12	(100)	Options
07-Apr-14	I IBC Socurition Australia 144	בוונבו ווונס הבוואמרואב	100 FF	(001,1)	Options
7 Apr 14	LIDE Cocurities Australia Ltd	341	160,110	(/2,233)	Orginary
1 2 2	Libe Securities Australia Lib	puy.	381	45	Ordinary
07-Apr-14	Ups securities Australia Ltd	Sell	056,85	(4,604)	Ordinary
-Apr-14	Ubs Securities Australia Ltd	Sell	11,895	(1,406)	Ordinary
5	Obs securities Australia LIU	llac	159'5	(899)	Ordinary
J/-Apr-14	UBS Securities Australia Ltd	Sell	48,214	(5,699)	Ordinary
1/-Apr-14	UBS Securities Australia Ltd	Sell	7,106	(840)	Ordinary
37-Apr-14	UBS Securities Australia Ltd	Buy	6,294	744	Ordinary
-Apr-14	UBS Securities Australia Ltd	Buy	41,843	4,946	Ordinary
-Apr-14	UBS Securities Australia Ltd	Buy	33	4	Ordinary
38-Apr-14	UBS AG Australia Branch	Transfer-In	A/N	1,495	Ordinary
38-Apr-14	UBS AG Australia Branch	Transfer-In	ΑN	10,979	Ordinary
18-Apr-14	UBS AG Australia Branch	Transfer-In	ΑN	1,084	Ordinary
38-Apr-14	UBS AG Australia Branch	Transfer-In	ΑN	2,849	Ordinary
8-Apr-14	UBS AG Australia Branch	Transfer-In	N/A	1,824	Ordinary
38-Apr-14	UBS AG Australia Branch	Sell	391,456	(46,079)	Ordinary
38-Apr-14	UBS AG London Branch	Stock borrowed	N/A	133	Ordinary
38-Apr-14	UBS AG London Branch	Stock borrowed	AN	5,333	Ordinary
38-Apr-14	UBS AG London Branch	Stock borrowed	ΑN	21,335	Ordinary
08-Apr-14	UBS AG London Branch	Stock borrowed	NA	294	Ordinary
38-Apr-14	UBS AG London Branch	Stock borrowed	A/N	11 507	Ordinary
38-Apr-14	UBS AG London Branch	Stock borrowed	A/N	13 334	Ordinary
38-Apr-14	UBS AG London Branch	Stock borrowed	AW	760	Ordinary
08-Apr-14	UBS Securities Australia I td	S _i ×	16.486	1 976	Coding
38-Apr-14	I IRC Countities Australia 14d	(A)	00+00	0/2/1	O'GILIBIY
20 Apr 14	Libe committee Australia Liu	Ana d	7//71	005,1	Ordinary
1 1 1	i De Comition Australia III	Ana S	304/1/8	114,087	Orginary
1 V V V V	LIBS Countition Australia I to	illac Series	955,950,1	(128,613)	Ordinary
00 Apr 14	Libe countries Australia Ltd	Jiac C	052,230	(14,500)	Orginary
10 Apr 14	Libe committee Australia Lid	Sons	44,750	200'5	Ordinary
08-Apr-14	LIBC Counties 110	Ctock horrowal	533,209	39,162	Ordinary
00 Apr 14	LIBS Sociation III	Stock bollowed	YA.	3,75	Orginary 0 i
- I- I- I- I- I- I- I- I- I- I- I- I- I-	יוסט סברתווופס רבר	Stock borrowed	¥2	104	Ordinary
11-104-	OBS Securities Australia Ltd	Enter Into Derivative U	AN.	(184)	Options
US-Apr-14	UBS Wealth Management Australia Ltd	Seil	1,689	(199)	Ordinary
08-Apr-14	UBS Securities Australia Ltd	Buy	5,455	638	Ordinary
08-Apr-14	Australia	- Xe	217,213	(25,405)	Ordinary
08-Apr-14	UBS Securities Australia Ltd	BUY	368	43	Ordinary
08-Apr- (4	UBS Securities Australia Ltd	я. Я	223,514	(26,142)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	∞	(1)	Ordinary
08-Apr-14	UBS Securities Australia Ltd	Buy	7,055	832	Ordinary
08-Apr-14	UBS Securities Australia Ltd	Buy	1,959	231	Ordinary
08-Apr-14	UBS Securities Australia Ltd	Buy	29,360	7,000	Ordinary
08-Apr-14	UBS Securities Australia Ltd	Buv	25.016	0100	
			200	056,2	Ordinary
08-Apr-14	UBS Securities Australia Ltd	Buv	382	45	Ordinary

08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 08-Apr-14 UBS Securities Australia Ltd Buy 09-Apr-14 UBS Securities Australia Ltd Buy 09-Apr-14 UBS Securities Australia Ltd Buy	2,603 29,680 170 23,532 2,408	307	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Secur	29,680 170 23,532 2,408	3,500	Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Secur	23,532 2,408		
UBS Securities Australia Ltd UBS Securities UBS Vealth UBAnagement Australia Ltd UBS Securities Australia Ltd	23,532	20	Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securit	2,408	2,775	Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securi		284	Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Secur	1.492	176	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	187	22	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	26.610	3 130	Ordingo
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UB	1 730	2000	Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securi	05,,	107	Ciditaly
UBS Securities Australia Ltd UBS Securities Australia Ltd	186,2	707	Cramary
UBS Securities Australia Ltid UBS Securities Australia Ltid	1,331	/5/	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,442	170	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	229	27	Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securit	1,187	140	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2.544	300	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3 155	372	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Secur	1 2/8	147	Ordingo
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Ltd UBS Securit	227 37	100	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS Securities Australia Ltd	10,00	1000	O Claimary
UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	201 000	7	Cigniary Cigniary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,130,492	255,45b	Ordinary
UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	4,462,500	(525,000)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	/16,51/	(1,920)	Ordinary
UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Ltd UBS Securities Australia Ltd	643,648	(75,000)	Ordinary
UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	6,787	(286)	Ordinary
UBS securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	25,915	3,000	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	788,773	91,539	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	777.717	(83.322)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	384 035	(44 738)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	45.943	(5 240)	Ordings
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	10,000	(010.0)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	291,102	(0C0'C)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	6,123	(717)	Ordinary
UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS Securities Australia It of UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia It of UBS Securities It of UBS S	N/A	103,484	Options
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	NA	83	Options
UBS Securities Australia Ltd UBS Gecurities Australia Ltd UBS Gecurities Australia Ltd UBS Gecurities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	N/A	40	Options
UBS Securities Australia Ltd UBS Global Assex Management UBS Weakib Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	ΥN	20	Options
UBS Global Asset Management Australia Ltd UBS Weakin Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	AM	20	Options
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	409 983	47.485	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd	111 465	10000	Ordings
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	55,10	000,21	6 6 6
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1/1/00	(0/C'/)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AC Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	687'5	(615)	Ordinary
UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securitiae Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	167,666	(19,496)	Ordinary
UBS Securities Australia It of UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia It of	215,000	(25,000)	Ordinary
UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. Securitian Branch UBS. Securitial Branch UBS. Securitial Branch UBS. Securitial Branch UBS. Securitial Ltd	198	(23)	Ordinary
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	221,534	(25.841)	Ordinary
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1.425,593	(166.289)	Ordinary
UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. AG Australia Branch UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd UBS. Securities Australia Ltd	221 524	(100,000)	Craining
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securitias Australia Ltc UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	+CC,132	(400,041)	Cicilialy
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securitiala Branch UBS Securitiala Branch UBS Securitiala Branch UBS Securitiala Lici	1,425,533	(697'001)	Ordinary
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	221,534	(25,841)	Ordinary
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1,425,593	(166,289)	Ordinary
UBS AG Australia Branch UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid	171,632	20,000	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	171,632	20,000	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	171 632	20.000	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	0360	(1 000)	Ordina
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	600,6	(050,1)	Oldling!
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	/3/,346	910,08	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	711,498	(82,959)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	197,558	(22,994)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	N/A	103,484	Options
UBS Securities Australia Ltd USS Wealth Management Australia Ltd USS Wealth Management Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd	Α/N	103.484	Options
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	AVA	9	Options
UBS Wealth Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	15.255	(1.778)	Ordinary
USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd	13 800	1,610	Ordinary.
UBS Securities Australia Ltd UBS Securities Australia Ltd	000000	00000	or all all y
UBS Securities Australia Ltd UBS Securities Australia Ltd	799,502	79,000	Ordinary
IUBS Securities Australia Ltd	1,493	174	Ordinary
	4,951	(577)	Ordinary
	177	(00)	Ordinary
Dia dipagnet can pose coo	7/1	(07)	Clourary

				2	
UBS Securities Australia Ltd	8.15	Buy	246	67	0.00
Securities Australia Ltd	8	Buy	25,016	2,950	Ordinary
UBS Securities Australia Ltd		Buy	54,230	6,395	Ordinary
ecurities Australia Ltd	ш	Buy	365	43	Ordinary
Securities Australia Ltd	В	Buy	4,321	509	Ordinary
UBS Securities Australia Ltd	В	Buy	9,415	1,109	Ordinary
UBS Securities Australia Ltd	8	Buy	66	-11	Ordinary
Securities Australia Ltd		Buy	19,555	2,306	Ordinary
Securities Australia Ltd	В	Buy	25,016	2.950	Ordinary
UBS Securities Australia Ltd		Buy	4 444	524	Ordinary
HRS Sequentities Australia 14d		D. W.	30 CC	109.0	Ordinan
DS Countries Academic Land		,	25,042	450,2	Countries
Decurities Australia Liu		ána .	2,544	OOC .	Ordinary
UBS securifies Australia Ltd	2	Buy	3/3	44	Ordinary
JBS Securities Australia Ltd	8	uy.	1,264	149	Ordinary
UBS Securities Australia Ltd	8	Buy	1,240	146	Ordinary
Securities Australia Ltd		Buv	730	86	Ordinary
Securities Australia Ltd		Billy	1 857	210	Ordinan
I IBC Con mition Australia 1+d			83	1	(paried
Securities Australia Ltd		had	80 00	0 000	Crainary
OBS Securities Australia Ltd	80	Buy	29,476	3,470	Ordinary
Securities Australia Ltd	8	Buy	12,372	1,459	Ordinary
UBS Securities Australia Ltd	8	Buy	1,365	191	Ordinary
UBS Securities Australia I td		Bliv	763	G	Ordinary
IBC Contrition Australia 14d		200	6614	7007	Ordinan
Securities Australia Ltd			1000	00/	Countries
Securities Australia Ltd	9	Buy	8,167	362	Ordinary
UBS Securities Australia Ltd	8	Buy	11,177	1,318	Ordinary
Securities Australia Ltd	8	Buy	068	105	Ordinary
UBS Securities Australia Ltd		Buv	212	25	Ordinary
Securities Australia 14d	a	2 40	001	3	Ordinary
DDD Decord lifes Australia Ltd		ens	500	3	Grandly
Securities Australia Ltd	20	Buy	4,211	496	Ordinary
UBS Securities Australia Ltd	8	Buy	1,630	192	Ordinary
Securities Australia Ltd	<u> </u>	Buy	13,136	1,549	Ordinary
JBS Securities Australia Ltd	8	Buv	17	2	Ordinary
JBS Securities Australia Ltd	8	Buv	1373	44	Ordinary
Secretary Australia	a	2.0	1169	101	Ordinary
100 Cocurition Australia (*d			210	90.	Ordinary
Canada Autor Canada		650	0 0	3 8	0.00
Dos seculties Australia Ltd		had	021,1	751	Orginary
UBS Securities Australia Ltd	2	á	087	33	Ordinary
JBS Securities Australia Ltd	<u>a</u>	Buy	3,158	372	Ordinary
JBS Securities Australia Ltd	9	Buy	3,770	444	Ordinary
JBS Securities Australia Ltd	8	Buv	3.277	386	Ordinary
IBS Socirities Australia 14d	0	a di	1687	551	Ordinan
Securities Australia Ltd		ń	100/1	100	Comission
UBS Securities Australia Ltd	8	3	3,800	447	Ordinary
Securities Australia Ltd	B	Buy	1,989	234	Ordinary
JBS Securities Australia Ltd	8	Buy	4,029	474	Ordinary
IBS Secritities Australia Ltd	8	Riny	408	87	Ordinan
		,	200	1	Color and
UBS Securities Australia Ltd	2	Buy	871	5	Ordinary
UBS Securities Australia Ltd	8	Buy	14,254	1,675	Ordinary
Securities Australia Ltd	B	Buv	2,389	281	Ordinary
UBS Securities Australia Ltd	ď	Rin	308	35	Ordinary
IDC Committee Assessed to the			4 222	2	, and a
Securities Australia Ltd	٥	buy	4,332	500	Ordinary
Securities Australia Ltd	В	Buy	502	29	Ordinary
UBS Securities Australia Ltd	æ	Buy	77	თ	Ordinary
Securities Australia Ltd	8	A	3.562	419	Ordinary
IBC Consisting Australia 144		9,11	264	10	Cardinan
ברתווונים שתפון פוופ בנת		<u></u>	107	- 600	2000
ODS Securities Australia Ltd		Ana	706,01	766'	Ordinary
UBS Securities Australia Ltd	ά	Buy	477	56	Ordinary
UBS Securities Australia Ltd	B.	Buy	2,553	300	Ordinary
UBS Securities Australia Ltd	añ.	Buy	229	27	Ordinary
JBS Securities Australia Ltd	æ	Buv	119	14	Ordinary
IBS Sociatistics Australia 144	0		20		Ordinan
Securities Adstralia Lid		, and	Co I	2 8	Ciciliaiy
UBS Securities Australia Ltd	201	BUY	50/	83	Ordinary
UBS Securities Australia Ltd	ā	Buy	3,506	413	Ordinary
UBS Securities Australia Ltd	ā	Buv	1.052	124	Ordinary
Securities Australia Ltd	a		7/15	08	Ordinan
ODS Seculities Australia Ltd	0	puy	740	8	5
Securities Australia Ltd	<u>m</u>	-			
		3.7	102	12	Ordinary

	UBS Securities Australia Ltd	Ę,	976	(301)	Ordinary
	or securities Australia Etc	-	CO7 C	(007)	
	35 Securities Australia I to	100	2,702	(432)	Ordinary
	SC Countries Australia Ltd	P	4,093	(1/5)	Ordinary
	DDS Securities Australia Ltd	III.	706,6	(618)	Ordinary
	UBS Securities Australia Ltd	Buy	3,134	364	Ordinary
	SS Seculines Australia Liu	áng	7,583	300	Ordinary
	DDS Securities Australia Lid	ans.	nac	G	Ordinary
	DD Securities Australia Ltd	buy.	3,134	304	Ordinary
	DES Securities Australia Ltd	and.	293	34	Ordinary
	Securities Australia Lid	no.	3,134	đ,	Ordinary
T	So Securities Australia LIO	euy euy	3,134	354	Ordinary
	DBS Securities Australia ctd	Buy	11,/53	1,365	Ordinary
1-14-0B	ss securities Australia Ltd	euy	5	-	Ordinary
Ť	JBS Securities Australia Ltd	Buy	21,125	2,445	Ordinary
Ť	JBS Securities Australia Ltd	Buy	202	24	Ordinary
7	JBS Securities Australia Ltd	Buy	3,145	364	Ordinary
_	JBS Securities Australia Ltd	Buy	3,940	456	Ordinary
7	JBS Securities Australia Ltd	Buy	9,213	1,062	Ordinary
0-Apr-14 UB	35 Securities Australia Ltd	Buy	3,160	364	Ordinary
-	JBS Securities Australia Ltd	Buy	3,160	364	Ordinary
0-Apr-14 UBS	35 Securities Australia Ltd	Buy	7,421	855	Ordinary
0-Apr-14 UBS	3S Securities Australia Ltd	Buy	5,442	627	Ordinary
Ť	JBS Securities Australia Ltd	Buy	3,160	364	Ordinary
	S Securities Australia Ltd	Buy	3,160	364	Ordinary
	UBS AG London Branch	Buy	338,234	39.649	Ordinary
1-Apr-14 UB	UBS AG London Branch	Sell	335.711	(39,649)	Ordinary
┢	JBS AG London Branch	Buy	41.782	4.936	Ordinary
Ī	UBS AG London Branch	la y	42 096	(4 036)	Ordinan
Γ	UBS Securities Australia I1d	Stock horrowed	V/N	0000 6	Ordinary
Ť	S Securities Australia I to	N. 1	2 003	25.1	Ording
T	S Securities Australia I to	Coll	1 700	1000	1000
1-Apr-14	100 Counties Australia 14d	1100	00/1	(2007)	Ordinary
T	IBC Countition Australia 14d	log.	210,000	141,121	Coldinary
Ť	IDC Countries Australia (14)	- C	10,014	(40,502)	Cidinary
T	Ups Securities Australia Ltd	buy 5-11	338,234	39,649	Ordinary
T	S Seculities Australia Ltd	100	44,030	(4,93b)	Ordinary
Ť	s securities Australia Ltd	Sell	324,138	(38,019)	Ordinary
Ť	UBS Securities Australia Ltd	Enter Into Derivative C	NA	000'009	Options
1	UBS Securities Australia Ltd	Enter Into Derivative (NA	(200)	Options
1-Apr-14 UBS	UBS Securities Australia Ltd	Enter Into Derivative (NVA	100	Options
1-Apr-14 UBS	UBS Wealth Management Australia Ltd	Buy	2,120	249	Ordinary
1-Apr-14 UBS	UBS Securities Australia Ltd	Buv	7,293	854	Ordinary
Ť		Buy	7 268	051	ordina.
1-Apr.14		, and	7,200	100	Guinary
Т	UBS Committee Australia Ltd	buy Dui:	40.00	8 2	Ordinary
T		euy	126,435	14,805	Ordinary
1	UBS Securities Australia Ltd	Jes	2,724	(319)	Ordinary
- 1	UBS Securities Australia Ltd	Buy	7,268	851	Ordinary
- 1	S Securities Australia Ltd	Buy	6,772	793	Ordinary
- 1		Sell	564	(99)	Ordinary
1-Apr-14 UBS		Buy	1,717	201	Ordinary
1-Apr-14 UBS	JBS Securities Australia Ltd	Buy	196	23	Ordinary
1-Apr-14 UBS	JBS Securities Australia Ltd	Buy	136,315	15,962	Ordinary
Ē	JBS Securities Australia Ltd	Buy	80.225	9.394	Ordinary
T	S Securities Australia Ltd	Riiv	6.832	SOO	Ordinan
t	IRS Sequerities Australia 11d	Buy	147 244	17.212	
1-40-14	IDC Committee Australia Ltd	,,,,,	107.044	216,7	Ordinaly
Ť	DC Contribut Australia Ind	, and	10,797	1,200	Ordinary
T	200 Secretary August August 200	(Po	100	207	Cramary
Ť	Socialities Australia Ltd	boy.	506	9 (Ordinary
T	UBS Securities Australia Ltd	any	/	7	Ordinary
11-Apr-14 UBS		Buy	34,075	3,990	Ordinary
1		Buy	2,904	340	Ordinary
Ť		Buy	12,998	1,522	Ordinary
┪	UBS Securities Australia Ltd	Buy	12,272	1,437	Ordinary
11-Apr-14 UBS	UBS Securities Australia Ltd	Buy	1,495	175	Ordinary
	S Securities Australia Ltd	Buy	956	112	Ordinary
Ī		Buv	2 639	300	Ordinary
Т	The substitute of the substitu	600	2,039	SOC .	Ciginaly

T		122			
T	UBS Securities Australia Ltd	Sell	5,277	(615)	Ordinary
T	JBS Securities Australia Ltd	gnà	2,128	248	Ordinary
0-Apr-14	OBS Securities Australia Lid	gng.	6,524	7//	Ordinary
Ť	UBS Securities Australia Ltd	III Sell	9,944	(1,159)	Ordinary
0-Apr-14	Obsider Adsirana Lid	Jac.	867 12	(66)	Ordinary
T	ARS Societies Australia I to	a d	756 73	(6/6/17)	Ordinaly
1	IBS Securities Australia Ind	200	12 013	0,000	Ordinary
Ť	UBS Securities Australia Ltd	Buv	4.290	2005	Ordinary
10-Apr-14 L	JBS Securities Australia Ltd	Sell	429	(20)	Ordinary
П	UBS Securities Australia Ltd	Self	883	(103)	Ordinary
	JBS Securities Australia Ltd	Sell	1,483	(173)	Ordinary
	UBS Securities Australia Ltd	Sell	1,191	(139)	Ordinary
	UBS Securities Australia Ltd	Sell	1,851	(216)	Ordinary
	JBS Securities Australia Ltd	Sell	2,924	(341)	Ordinary
0-Apr-14 U	UBS Securities Australia Ltd	Sell	3,085	(360)	Ordinary
П	UBS Securities Australia Ltd	Sell	1,551	(181)	Ordinary
7	JBS Securities Australia Ltd	Sell	1,140	(133)	Ordinary
7	UBS Securities Australia Ltd	Sell	2,468	(288)	Ordinary
1	JBS Securities Australia Ltd	les.	11,929	(1,392)	Ordinary
T	UBS Securities Australia Ltd	9	929	(73)	Ordinary
Ť	UBS Securities Australia Ltd		2,007	(234)	Ordinary
Ť	JBS Securities Australia Ltd	B	1,208	(141)	Ordinary
0-Apr-14	UBS Securities Australia Ltd	Neil Neil	119,1	(522)	Ordinary
T	IBS Securities Australia Ltd	Sel.	D (2)	(1)	Ordinary
10-Apr-14	IRS Sequities Australia 11d	5 0	350	(15)	Ordinary
T	JBS Securities Australia Ltd	las.	1 028	(120)	Ordinary
T	UBS Securities Australia I td	25	759	(18)	Ordinary
Ť	UBS Securities Australia Ltd	Sell	7.045	(822)	Ordinary
ı	BS Securities Australia Ltd	les	2,717	(317)	Ordinary
	BS Securities Australia Ltd	Sell	5,459	(637)	Ordinary
10-Apr-14 U	UBS Securities Australia Ltd	Sell	3,006	(351)	Ordinary
	JBS Securities Australia Ltd	Sell	2,013	(532)	Ordinary
Ť	JBS Securities Australia Ltd	Sell	1,585	(185)	Ordinary
	JBS Securities Australia Ltd	Sell	958	(100)	Ordinary
10-Apr-14 U	UBS Securities Australia Ltd	Sell	3,407	(366)	Ordinary
	IBS Securities Australia Ltd	Sell	1,225	(143)	Ordinary
0-Apr-14 U	BS Securities Australia Ltd	Sell	5,267	(615)	Ordinary
- 1	BS Securities Australia Ltd	Seil	5,841	(682)	Ordinary
10-Apr-14 U	IBS Securities Australia Ltd	Sell	103	(12)	Ordinary
- 1	BS Securities Australia Ltd	Sell	848	(66)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	229	(62)	Ordinary
- 1	BS Securities Australia Ltd	Sell	1,715	(200)	Ordinary
	UBS Securities Australia Ltd	Sell	1,003	(117)	Ordinary
Ī	UBS Securities Australia Ltd	Seff	2,298	(368)	Ordinary
	UBS Securities Australia Ltd	Selí	2,341	(273)	Ordinary
	JBS Securities Australia Ltd	Sell	1,235	(144)	Ordinary
10-Apr-14 U	JBS Securities Australia Ltd	Sell	755	(88)	Ordinary
Ĩ	JBS Securities Australia Ltd	Zell	1,715	(200)	Ordinary
_	JBS Securities Australia Ltd	Sell	696	(113)	Ordinary
	UBS Securities Australia Ltd	Sell	1,982	(231)	Ordinary
Ē	UBS Securities Australia Ltd	Sell	1,389	(162)	Ordinary
Ē	UBS Securities Australia Ltd	Sell	1,715	(200)	Ordinary
Ī	BS Securities Australia Ltd	Sell	763	(68)	Ordinary
Г	JBS Securities Australia Ltd	Self	2,324	(271)	Ordinary
0-Apr-14 U	UBS Securities Australia Ltd	lleS	1,355	(158)	Ordinary
f	JBS Securities Australia Ltd	ll-9S	1,431	(167)	Ordinary
Ť	JBS Securities Australia Ltd	Sell	2,778	(324)	Ordinary
Ē	JBS Securities Australia Ltd	Sell	935	(109)	Ordinary
П	BS Securities Australia Ltd	Self	420	(49)	Ordinary
	UBS Securities Australia Ltd	Seli	4,611	(238)	Ordinary
	UBS Securities Australia Ltd	Sell	102'9	(781)	Ordinary
-Apr-14 Ui	8S Securities Australia Ltd	Sell	7,027	(820)	Ordinary
Γ	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				
10-Apr-14	UBS Securities Australia Ltd	Sell	3.137	(398)	Ordinary

1-122-14	The second secon	_	,	2	Ordinary
4.Apr.14	UBS Securities Australia Ltd	in a	0	-	
14.Apr.14	LIBS Societies Australia Lid	buy Buy	8 100	- 000	Ordinary
Apr. 14	I IBC Societies Australia (4d	boy.	22,051	3,420	Ordinary
4.00-14	Libs Sociation Australia 144	an)	101/2	4,407	Orginary
4-Apr-14	LIBC Counties Australia 14d	bus,	7,30	60	Clumary
4-Anr-14	LIRS Securities Australia 144	Di N	10 430	760 1	Cidinaly
4-Apr-14	I IRC Convision Australia 144	600	0.770	7671	Clorinary
-Anr-14	1 RS Securities Australia 11d	, and	582	200	Ordinary
4-Apr-14	UBS Securities Australia Itd	N N	1 130	13.4	Ordinary
4-Apr-14	I BC Courties Australia 144	(Sala)	127.15	127.0	Ordinal y
4-Anr-14	LIRS Societies Australia 11d	Billy	157,15	2,704	Ordinary
4-Apr-14	I IRC Securities Australia 14d	Buy	700 3	747	Ordinary
4-Anr-14	I IRC Countries Australia 14d	Dony	1 550	100	Garday
4-Anr-14	I IRS Societies Australia 14d	Britis	999 31	1 077	Cidinaly
1	TIPE Consistent Australia Ltd	Buy	000,01	(/6,1	Ordinary
4 4 4 4 4 4 4	Upo Securities Australia Ltd	anà	0,323	720	Ordinary
4-4-01-14	Upo certaines Australia Ltd	gnà	4,133	552	Ordinary
4-ADI-14	Upo Securités Australia Ltd	Buy	676	35	Ordinary
4-ADI-14	Obsider Australia Ltd	Buy	089'51	098,1	Ordinary
4-AD-14	Upp Securities Australia Ltd	gny	33,880	4,000	Ordinary
2-Apr-14	USS AG AUstralia Branch	Sell	60,495	(7,149)	Ordinary
101	OBS AG LOTIGOT BIATION	liac	717'966	(17,986)	Ordinary
2-ADI-14	UBS Att London Branch	gnà	955,684	112,986	Ordinary
5-Apr-14	UBS Securities Australia Ltd	100	097'59	(6,291)	Ordinary
2-ADI-14	UBS Securities Australia Ltd	Buy	391,010	46,144	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Sell	281,108	(33,166)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	956,212	(112,986)	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Buy	29,435	3,500	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Sell	254,573	(30,069)	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Buy	49,736	5,887	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Sell	174,992	(20,649)	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative C	NA	174	Options
5-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (WA	20	Options
-Apr-14	UBS Global Asset Management	Buy	36,778	4,351	Ordinary
5-Apr-14	UBS Global Asset Management	Sell	2,957	(320)	Ordinary
5-Apr-14	UBS Wealth Management Australia Ltd	Buy	1,071	478	Ordinary
5-Apr-14	UBS Wealth Management Australia Ltd	Buy	8,805	1,034	Ordinary
-Apr-14	UBS Securities Australia Ltd	- Sel	7,968	(843)	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Jes e	79,795	(9.384)	Ordinary
5-Apr-14	I IRC Securities Australia 144	100	100	(4.1)	Ordings
5-Anr-14	I IRS Securities Australia 14d	3	75.080	(890 C)	Ordinary
5-Apr-14	I IRC Coourities Australia 14d		000,02	(005/2)	Ordingly
1	Opp Securities Australia Liu	ijac	200,5	(0/0)	Ordinary
2-Apr-14	UBS Securities Australia Ltd	lle c	32,558	(3,853)	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Sell	194	(23)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sel	6,861	(812)	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Sell	33,920	(4,000)	Ordinary
6-Apr-14	UBS AG Australia Branch	Seli	288'62	(9,286)	Ordinary
16-Apr-14	UBS AG Australia Branch	Seli	859,384	(99,515)	Ordinary
6-Apr-14	UBS AG Australia Branch	Buy	12,620	1,458	Ordinary
6-Apr-14	UBS AG Australia Branch	Seil	425,616	(20,000)	Ordinary
6-Apr-14	UBS AG Australia Branch	Sell	432,773	(20'000)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	3,127	(364)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Buy	666,917	77,236	Ordinary
-Apr-14	UBS Securities Australia Ltd	Seil	543,498	(63,004)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	10,763	(1,250)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	42,266	(4,915)	Ordinary
16-Apr-14	UBS Securities Australia Ltd	Sell	23,403	(2,700)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Buy	8,404	176	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	21,500	Options
16-Apr-14	UBS Securities Australia Ltd		ΝA	388,400	Options
6-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	388,400	Options
6-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	300	Options
6-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (AN	85	Options
6-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	200	Options
16-Anr-14					
	It IBS Securities Australia Ltd	Friter Into Derivative d		157	Ontions

1-Apr-14	HV Verillation Assets 12				Ordinan
	כמי אבייו איבי שמיו שום דומ	suy	122,8/4	14,388	Cicinaly
1-Apr-14	UBS Securities Australia Ltd	Buy	17,635	2,065	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Buy	116,409	13,631	Ordinary
11-Apr-14	UBS Securities Australia Ltd	suy	890'8/	8,556	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Sell	34,120	(4,000)	Ordinary
11-Apr-14	UBS Securities Australia Ltd	Buy	65	_	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Buy	4,619	244	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Buy	4,899	577	Ordinary
1-Api-14	UBS Securities Australia Ltd	anà	975	79	Ordinary
+ id	OBS Securities Australia Ltd	ony.	y 10,4	244	Ordinary
1-MDI-14	UDD Securities Australia Liu	euy	8/0,01	1,18/	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Buy	8,475	000,1	Ordinary
Apr-14	UBS Securities Australia Ltd	Buy	2,509	296	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Buy	831	86	Ordinary
11-Apr-14	UBS Securities Australia Ltd	Buy	12,720	1,500	Ordinary
14-Apr-14	UBS AG Australia Branch	Buy	422,464	20,000	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Self	3,316	(393)	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Xet Xet	75,352	(3,000)	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	267,534	31,754	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	305,874	(36,312)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	42,050	2,000	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Sell	453,258	(53,729)	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Sell	4,579	(545)	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	28,915	3,430	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	31,688	3,759	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	5,311	630	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	13,977	1,658	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	19,549	2,319	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	157,700	18,707	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	72,827	8,639	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	1,973	234	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	24,557	2,913	Ordinary
Apr-14	UBS Securities Australia Ltd	Buy	1,020	121	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	77,345	9,175	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	266,026	31,557	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	40,936	4,856	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	79,562	9,438	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	1,526	181	Ordinary
-Apr-14	UBS Securities Australia Ltd	Buy	19.549	2,319	Ordinary
14-Anr-14	I IRS Securities Australia 11d	Bir	71 301	8 A58	Ordinan
14 Apr 14	Jibo Connition Australia (+d	623	730 051	20,000	Oldinary Oldinary
A-Apr-14	1 DC Countities Australia 14d	Bush	1 115	150	Ordinary
A - 14	Ling complete Australia Ltd	Buy	1,410	00 5	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	187'1	751	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	89,763	10,648	Ordinary
-Apr-14	UBS Securities Australia Ltd	Buy	146,598	17,390	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	4,881	579	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	72,827	8,639	Ordinary
ı	UBS Securities Australia Ltd	Buy	12,527	1,486	Ordinary
	UBS Securities Australia Ltd	Buy	72,827	8,639	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	79,276	9,404	Ordinary
1	UBS Securities Australia Ltd	BÇ	2,596	308	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buv	15.410	1 828	Ordinary
14-Apr-14	I IRS Securities Australia I to	Birk	90.445	10.720	Ordinan
Ł	LIBS Sequirities Australia Ltd	Buy	99,00	11 744	Ordinary
	LIBS Sequitities Australia Ltd	Buv	20,603	2 444	Ordinary
14-Anr-14	I IBS Securities Australia I to	Birk	161 342	10 130	Ordinary
1	# IRC Securities Australia 14d	Briv	7 7 2 7	560	Ordinary
14-Apr-14	18S Securities Australia 11d	Bris	16.880	2,000	Ordings
Apr.14	LIBS Securities Australia Ltd	9.10	0000	2,000	Ordinary
14-Apr-14	I IRS Securities Australia 14d	Rin	3 933	2,000	Ordinary
4-Apr-14	I IBS Societies Australia 14d	No.	205,00	700	Ordinary
1100	LIDE Contribute Australia Ltd	ond,	1/5/62	3,400	Ordinaly
4-7401-14	Tipe Securities Australia Ltd	ing a	750,051	191,01	Ordinary
4-ADI-14	UBS Securities Australia Ltd	Buy	00/	3.	Orginary
4-Apr-14	UBS Securities Australia Ltd	Buy	20	-	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Buy	514	61	Ordinary
14-Apr-14	UBS Securities Australia Ltd	Buy	808	96	Ordinary
11		i d	031 57	000	

7-Apr-14	OBS Securities Australia Ltd	iik.	/74//50	(75.354)	Crainary
-	ILIBA Securities Australia i to	N. S.	115 966	13 402	Ordinan
Apr-14	UBS Securities Australia Ltd	les les	42.591	(4,923)	Ordinary
7-Apr-14	UBS Securities Australia I td	و	FOT 87A	(49.485)	Ordinary Ordinary
7-Anr-14	I IRC Securities Australia I to	200	550,024	(10,100)	Oddinal y
7-Apr-14	I IBS Constition Australia Ltd	600	27970	900	Courielly
Apr-14	LIBS Global Asset Management	B. S.	2,2,2,2,2 2,2,5,0,2	375	Ordinary
7-Anr.14	BC Counties Australia 14d	000	1 524	1961	O G
Apr. 14	186 Securities Australia 14d	100	000.3	(0/1/	i i
7-Apr-14	UBS Securities Australia I td	9	328	(38)	ordinary.
7-Apr-14	UBS Securities Australia Ltd	N N	70 676	92,0	Ordinary
7-Apr-14	LIBS Securities Australia 11d	(S) (S)	1 356	157	Ordinal y
Apr-14	I IBC Counties Australia 14d	600	C20 0E	00,00	Oldinary Oldinary
7-Apr.14	I IBC Contrition Australia 14d	100	10,007	601,0	Cidinary
14	Upo Securities Australia Ltd	nec l	/+	(7)	Ordinary
4-1d-14	UBS Securities Australia Ltd	Sel!	806	(104)	Ordinary
/-Apr-14	UBS Securities Australia Ltd	Self	454	(52)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	454	(52)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	454	(52)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Seli	454	(52)	Ordinary
Apr-14	UBS Securities Australia Ltd	Sell	79	(6)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	454	(52)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Seli	79	(6)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Self	716	(82)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	454	(25)	Ordinary
Apr-14	UBS Securities Australia Ltd	Sell	1,013	(116)	Ordinary
Apr-14	UBS Securities Australia Ltd	Self	742	(82)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Seli	26	(3)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	096	(110)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	σ	ε	Ordinary
-Apr-14	UBS Securities Australia I td	3	1 467	(168)	Ordinary
Apr. 14	I IBC Cocurties Australia Ltd	100	700	(110)	, and
	The Committee Australia Lite	100	*00'	(011)	Ordinary
1 2 2	Opp Securities Australia Lid	i i i	454	(75)	Ordinary
1	Ubs securities Australia Ltd	Seil	1,951	(338)	Ordinary
Apr-14	UBS Securities Australia Ltd	Seil	24,881	(2,850)	Ordinary
/-Apr-14	UBS Securities Australia Ltd	Seil	879'7	(301)	Ordinary
/-Apr-14	UBS Securities Australia Ltd	Sell	349	(40)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	7,063	(808)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	8,136	(632)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	454	(52)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	646	(74)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	lleS	757	(52)	Ordinary
Apr.14	IIR Securities Australia I to	= 0	753 66	(20)	Ordinary.
1 4 4 4 4	Upo con mices Adamada Lid	nac .	/50/57	(4,095)	Orginary
1 1 1	Obs securities Adstratia Ltd	i a	#C#	(75)	Orginary
/-ADr-14	UBS Securities Australia Ltd	Sell	160,1	(125)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Self	454	(52)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	1,065	(122)	Ordinary
Apr-14	UBS Securities Australia Ltd	Sell	454	(25)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	454	(25)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	454	(52)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	454	(52)	Ordinary
7-Apr-14	UBS Securities Australia 11d	روا	454	(52)	Ordinary
Apr-14	UBS Securities Australia 11d	Sell	454	(52)	Ordinary
7-Apr-14	LIBS Securities Australia Ltd	lo Co	454	(52)	Ordinary
Apr. 14	I IDC Counties Australia 144		454	(20)	Countries of
1 Apr 14	Upo Securities Australia Ltd)eil	404	(25)	Ordinary
1 1 4	LIDE Commission Australia ted	ne c	14C	(20)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	la la	404	(75)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	les.	3,117	(357)	Ordinary
/-Apr-14	UBS Securities Australia Ltd	lles.	716	(105)	Ordinary
/-Apr-14	UBS Securities Australia Ltd	Sell	533	(61)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	1,248	(143)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	175	(20)	Ordinary
7-Apr-14	UBS Securities Australia Ltd	Sell	1,956	(224)	Ordinary
2-Apr-14	UBS AG Australia Branch	Buy	141,189	16,250	Ordinary
Apr-14	UBS AG Australia Branch	Sell	871.291	(100,000)	Ordinary
22-Apr-14				********	
1	ILIRS Securities Australia Ltd	74.00	1767	203	Ordinan

					200
16-Apr-14	UBS Securities Australia Ltd	Buy	2.073.600	240.000	Ordinary
16-Apr-14	UBS Securities Australia Ltd	Sell	43,200	(5,000)	Ordinary
16-Apr-14	UBS Securities Australia Ltd	Self	622,564	(72,056)	Ordinary
16-Apr-14	UBS Securities Australia Ltd	J	2,592	(300)	Ordinary
16-Apr-14	UBS Securities Australia Ltd	lleS.	86.400	(10,000)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	lles	117,210	(13,566)	Ordinary
16-Apr-14	UBS Securities Australia Ltd	lley.	218,195	(25,254)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	323,456	(37,437)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	31,372	(3,631)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	693,541	(80,271)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	2,324	(592)	Ordinary
16-Apr-14	UBS Securities Australia Ltd	Sell	12,951	(1,499)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Self	14,342	(1,660)	Ordinary
	UBS Securities Australia Ltd	Sell	10,714	(1,240)	Ordinary
	UBS Securities Australia Ltd	Sell	3,482	(403)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	13,185	(1,526)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	106,963	(12,380)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	173	(20)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Seil	20,028	(2,318)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Set	25,730	(2,978)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Seli	118,299	(13,692)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	20,028	(2,318)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	10,100	(1,169)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	6,929	(802)	Ordinary
15-Apr-14	UBS Securities Australia Ltd	Sell	20,028	(2,318)	Ordinary
b-Apr-14	UBS Securities Australia Ltd	lle S	8,035	(930)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	- Ne	2,825	(37)	Ordinary
0-Apr-14	Ups Securities Australia Ltd	<u> </u>	8/	6	Ordinary
0-Api-14	UBS Securities Australia Ltd	II-J	20,028	(2,518)	Ordinary
4-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1	UBS Securities Australia Ltd	130	35,751	(2,749)	Ordinary
6 Apr 14	UBS Securities Australia Ltd	Jac J	323,130	(37,400)	Ordinary
6-Apr-14	185 Securities Australia 114	Sail	677.0	(101)	Ordinary
6-Apr-14	UBS Securities Australia 14d	3	27775	(15)	Ordinary
6-Anr-14	UBS Securities Australia 14d	lle2	80000	(2318)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	127,656	(14.775)	Ordinary
6-Apr-14	UBS Securities Australia I td	95	3 905	(452)	Ordinan
6-Apr-14	UBS Securities Australia Ltd	lles	2.004	(250)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Jes.	3.479	(401)	Ordinary
6-Apr-14	UBS Securities Australia I td	و	1 795	(202)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sett	642	(74)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	26	0	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	36	00	Ordinary
16-Apr-14	UBS Securities Australia Ltd	Sell	1,209	(140)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	1,554	(180)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	958	111	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	2.045	(237)	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Sell	2,567	(300)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	8,153	(623)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	8,365	(883)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	2,698	(317)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	13,148	(1,545)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	85	(10)	Ordinary
16-Apr-14	UBS Securities Australia Ltd	Sell	6,629	(277)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	ll ₃ S	2,519	(296)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	2,572	(302)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	3,344	(393)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sell	5,072	(965)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	ll se	4,743	(557)	Ordinary
16-Apr-14	UBS Securities Australia Ltd		2,282	(268)	Ordinary
6-Apr-14	UBS Securities Australia Ltd	Sel	111	(13)	Ordinary
7-Apr-14	UBS AG Australia Branch	Sell	31,406	(3,633)	Ordinary
7-Apr-14	UBS AG Australia Branch	Seil	146,799	(12,000)	Ordinary
7-Apr-14	UBS AG Australia Branch	Sell	69,082	(8,000)	Ordinary
17-Apr-14	UBS AG Australia Branch	Sell	171,919	(19,816)	Ordinary
		-	0000		

2,643 (449) 2,643 (449) 1,1515 (1,116) 1,066 (115) 1,066 (115) 1,071 (1,62) 1,071 (1,62) 1,072 (1,169) 1,073 (1,169) 1,073 (1,169) 1,073 (1,169) 1,073 (1,169) 1,074 (1,169) 1,075 (1,16		A DC Care mittee A control and		0000		
USS Securities Australia Ltd Sell 1,516 (1,316)	23-Apr-14	UBS Securities Australia I td	la S	2,523	(302)	Ordinary
UBS Securities Australia Ltd Sell 1,006 (11) UBS Securities Australia Ltd Sell 5,005 (107) UBS Securities Australia Ltd Sell 5,005 (107) UBS Securities Australia Ltd Sell 10,006 (117) UBS Securities Australia Ltd Sell 10,006 (107) UBS Securities Australia Ltd Sell 10,006 (100) UBS Securities Australia Ltd Sell 10,006 (100) UBS Securities Australia Ltd Sell 10,006 (100) UBS Securities Australia Ltd Sell 10,007 (100) UBS Securities Au	23-Apr-14	UBS Securities Australia 11d	===	11 515	(1 316)	C C C
Big Securities Australia Ltd Self 1,000	23-Apr-14	UBS Securities Australia 11d		253	(12)	Ordinary
1985 Securities Australia Ltd Sell 1987 1988 198	200	Laboration A continuous 2001	11-0	466	10	Couries Couries
March Marc	23-Api-14	Upo Securities Australia Lid	Jac.	900,1	(611)	Ordinary
USS securities Australia Ltd Sell 19,378 1,1850	1	CDD Decollities Australia Ltd	100	966	(101)	Ordinary
UBS Securities Australia Ltd Sell 9.92 (1.06) UBS Securities Australia Ltd Sell 9.92 (1.06) UBS Securities Australia Ltd Sell 9.92 (1.06) UBS Securities Australia Ltd Sell 9.93 (1.06) UBS Securities Australia Ltd Sell 7.276 (60) UBS Securities Australia Ltd Sell 7.276 (1.078) UBS Securities Australia Ltd Sell 41,529 (1.078) UBS Securities Australia Ltd Sell 41,529 (1.08) UBS Securities Australia Ltd Sell 3,229 (1.09) UBS Securities Australia Ltd Sell 3,278 (1.09) UBS Securities Australia Ltd Sell 3,278 (1.09) UBS Securities	23-Apr-14	UBS Securities Australia Ltd	Sell	5,075	(280)	Ordinary
UBS Securities Australia Ltd Sell 934 (10) UBS Securities Australia Ltd Sell 13349 (2,097) UBS Securities Australia Ltd Sell 5,246 (600) UBS Securities Australia Ltd Sell 1,015 (1,018) UBS Securities Australia Ltd Sell 3,023 (1,018) UBS Securities Australia Ltd Sell 3,033 (1,018) UBS Securities Australia Ltd Sell 3,033 (1,019) UBS Securities Australia Ltd Sell 3,033 (1,019) UBS Securities Australia Ltd Sell 3,033 (3,018) UBS Securities Australia Ltd Sell 1,023 (1,019) UBS Securities Australia Ltd Sell 1,024 (1,021) UBS S	23-Apr-14	UBS Securities Australia Ltd	Sell	10,378	(1,186)	Ordinary
UBS Securities Australia Ltd Sell 1849 (4,6) BNB Securities Australia Ltd Sell 18,296 (6,0) BNB Securities Australia Ltd Sell 7,276 (6,0) BNB Securities Australia Ltd Sell 1,6,15 (1,16) BNB Securities Australia Ltd Sell 1,0,15 (1,16) BNB Securities Australia Ltd Sell 1,0,15 (1,16) BNB Securities Australia Ltd Sell 1,0,15 (1,16) BNB Securities Australia Ltd Sell 2,20 (2,66) BNB Securities Australia Ltd Sell 3,0,23 (4,124) BNB Securities Australia Ltd Sell 3,0,0 (4,124) BNB Securities Australia Ltd Sell 3,0,0 (4,124) BNB Securities Australia Ltd Sell 3,0,0 (4,124) BN	73-Apr-14	UBS Securities Australia Ltd	Seli	928	(106)	Ordinary
UBS Securities Australia Ltd Sell 5,349 (2,09) UBS Securities Australia Ltd Sell 5,249 (2,09) UBS Securities Australia Ltd Sell 1,015 (1,678) UBS Securities Australia Ltd Sell 4,528 (4,746) UBS Securities Australia Ltd Sell 4,528 (4,746) UBS Securities Australia Ltd Sell 4,528 (4,746) UBS Securities Australia Ltd Sell 3,723 (4,746) UBS Securities Australia Ltd Sell 3,723 (4,746) UBS Securities Australia Ltd Sell 3,623 (7,77) UBS Securities Australia Ltd Sell 3,623 (4,746) UBS Securities Australia Ltd Sell 3,63 (4,124) UBS Securities Australia Ltd Sell 3,73 (4,124) UBS Se	23-Apr-14	UBS Securities Australia Ltd	Sell	394	(42)	Ordinary
UBS Securities Australia Ltd Sell Sell Sel Col UBS Securities Australia Ltd Sell Sell Col UBS Securities Australia Ltd Sell Sell Col UBS Securities Australia Ltd Sell Col UBS Securities Australia Ltd Sell Col UBS Securities Australia Ltd Sell	23-Apr-14	UBS Securities Australia Ltd	Self	18,349	(2,097)	Ordinary
UBS Scorriles Australia Ltd Sell 79 (10) UBS Scorriles Australia Ltd Sell 1,678 (1,678) UBS Scorriles Australia Ltd Sell 1,015 (1,678) UBS Scorriles Australia Ltd Sell 1,015 (1,678) UBS Scorriles Australia Ltd Sell 1,015 (1,678) UBS Scorriles Australia Ltd Sell 8,7721 (4,311) UBS Scorriles Australia Ltd Sell 8,7721 (4,311) UBS Scorriles Australia Ltd Sell 3,623 (1,549) UBS Scorriles Australia Ltd Sell 3,721 (4,311) UBS Scorriles Australia Ltd Sell 3,639 (3,98) UBS Scorriles Australia Ltd Sell 3,639 (4,19) UBS Scorriles Australia Ltd Sell 3,539 (4,19) UBS Scorriles Australia Ltd Sell 3,539 (4,19) UBS Scorriles Australia Ltd Sell 3,539 (4,19) UBS Scorriles Australia Ltd Sell 3,540 (5,20) UBS Scorriles Australia	23-Apr-14	UBS Securities Australia Ltd	Self	5,276	(603)	Ordinary
UBS Securities Australia Ltd Sell 14,691 (1,698) UBS Securities Australia Ltd Sell 4,523 (1,698) UBS Securities Australia Ltd Sell 1,515 (1,124) UBS Securities Australia Ltd Sell 1,517 (1,124) UBS Securities Australia Ltd Sell 3,229 (4,426) UBS Securities Australia Ltd Sell 3,229 (1,124) UBS Securities Australia Ltd Sell 3,529 (4,123) UBS Securities Australia Ltd Sell 3,529 (1,124) UBS Securities Australia Ltd Sell 1,549 (1,124) UBS Securities Australia Ltd Sell 1,529 (1,124) UBS Securities Australia Ltd Sell 7,250 (1,124) UBS Securities Australia Ltd Sell 1,269 (1,124) UBS Securities Australia Ltd Sell 1,269 (1,124) UBS Securities Australia Ltd Sell 1,269 (1,124) UBS Securities Australia Ltd Sell 1,260 (1,109) <	23-Apr-14	UBS Securities Australia Ltd	Sell	79	6	Ordinary
USS Securities Australia Ltd Sell 4,1,691 (1,678) UBS Securities Australia Ltd Sell 4,1,28 (4,1,46) UBS Securities Australia Ltd Sell 1,015 (4,1,46) UBS Securities Australia Ltd Sell 3,229 (4,31) UBS Securities Australia Ltd Sell 3,229 (4,31) UBS Securities Australia Ltd Sell 3,229 (4,31) UBS Securities Australia Ltd Sell 3,529 (4,31) UBS Securities Australia Ltd Sell 3,529 (4,31) UBS Securities Australia Ltd Sell 3,539 (1,169) UBS Securities Australia Ltd Sell 1,740 (1,169) UBS Securities Australia Ltd Sell 3,539 (1,169) UBS Securities Australia Ltd Sell 1,750 (1,169) UBS Securities Australia Ltd Sell 1,750 (1,09) UBS Securities Australia Ltd Sell 1,750 (1,09) UBS Securities Australia Ltd Sell 1,750 (1,00) U	23-Apr-14	UBS Securities Australia Ltd	Self	919	(105)	Ordinary
USS Securities Australia Ltd Sell 4,15,26 (4,16) UBS Securities Australia Ltd Sell 1,51,58 (4,124) UBS Securities Australia Ltd Sell 1,51,58 (4,124) UBS Securities Australia Ltd Sell 3,529 (4,124) UBS Securities Australia Ltd Sell 3,529 (4,426) UBS Securities Australia Ltd Sell 3,529 (4,124) UBS Securities Australia Ltd Sell 3,529 (4,124) UBS Securities Australia Ltd Sell 7,250 (8,48) UBS Securities Australia Ltd Sell 7,250 (8,48) UBS Securities Australia Ltd Sell 1,26 (1,109) UBS Securities Australia Ltd Sell 1,26 (1,109) UBS Securities Australia Ltd Sell 1,26 (1,109) <t< td=""><td>23-Apr-14</td><td>BS Securities Australia 14d</td><td></td><td>14 601</td><td>(2017</td><td>Coloriary</td></t<>	23-Apr-14	BS Securities Australia 14d		14 601	(2017	Coloriary
UBS Securities Australia Ltd Sell 41,528 (4,746) UBS Securities Australia Ltd Sell 1,015 (1,169) UBS Securities Australia Ltd Sell 3,229 (4,246) UBS Securities Australia Ltd Sell 3,229 (4,511) UBS Securities Australia Ltd Sell 3,229 (4,511) UBS Securities Australia Ltd Sell 3,229 (4,4311) UBS Securities Australia Ltd Sell 3,229 (4,1311) UBS Securities Australia Ltd Sell 3,229 (4,1311) UBS Securities Australia Ltd Sell 1,520 (1,169) UBS Securities Australia Ltd Sell 1,220 (1,160) UBS Securities Australia Ltd Sell 1,220 (1,160)	20 Apr 14	Late Countries According Ltd	Jac.	14,091	(9/0'1)	Ordinary
UBS Securities Autorials Ltd Sell 1,12,88 (4,124) UBS Securities Autorials Ltd Sell 1,528 (4,124) UBS Securities Autorials Ltd Sell 2,229 (266) UBS Securities Autorials Ltd Sell 3,727 (4,124) UBS Securities Autorials Ltd Sell 3,229 (7,124) UBS Securities Autorials Ltd Sell 3,229 (7,124) UBS Securities Autorials Ltd Sell 1,549 (4,123) UBS Securities Autorials Ltd Sell 1,249 (1,134) UBS Securities Autorials Ltd Sell 1,269 (1,134) UBS Securities Autorials Ltd Sell 1,269 (1,134) UBS Securities Autorials Ltd Sell 1,269 (1,134) UBS Securities Autorials Ltd Sell 1,273 (4,130) UBS Securities Autorials Ltd Sell 1,265 (1,00) UBS Securities Autorials Ltd Sell 1,265 (1,00) UBS Securities Autorials Ltd Sell 1,267 (1,00) U	41-10-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	ODS SECULINES AUSTRAINE LIG	Sell	53	(q)	Orginary
UBS Securities Australia Ltd Sell 1,015 (1,134) (968) UBS Securities Australia Ltd Sell 3,329 (2,69) UBS Securities Australia Ltd Sell 36,083 (4,131) UBS Securities Australia Ltd Sell 37,721 (4,311) UBS Securities Australia Ltd Sell 10,229 (1,169) UBS Securities Australia Ltd Sell 10,229 (1,169) UBS Securities Australia Ltd Sell 10,229 (1,169) UBS Securities Australia Ltd Sell 1,206 (1,219) UBS Securities Australia Ltd Sell 1,206 (1,209) UBS Securities Australia Ltd Sell 1,206 (1,400) UBS Securities Australia Ltd Sell 1,206 (1,400) UBS Securities Australia Ltd Sell 1,206 (1,500)	43-Apr-14	UBS Securities Australia Ltd	Sell	41,528	(4,/46)	Ordinary
UBS Securities Australia Ltd Sell 8,470 (366) UBS Securities Australia Ltd Sell 2,095 (4,124) UBS Securities Australia Ltd Sell 3,723 (4,4124) UBS Securities Australia Ltd Sell 3,723 (4,439) UBS Securities Australia Ltd Sell 3,623 (1,169) UBS Securities Australia Ltd Sell 1,622 (1,169) UBS Securities Australia Ltd Sell 1,622 (1,169) UBS Securities Australia Ltd Sell 1,666 (1,169) UBS Securities Australia Ltd Sell 1,666 (1,169) UBS Securities Australia Ltd Sell 1,740 (1848) UBS Securities Australia Ltd Sell 1,740 (1848) UBS Securities Australia Ltd Sell 1,740 (1848) UBS Securities Australia Ltd Sell 1,750 (140) UBS Securities Australia Ltd Sell 1,750 (140) UBS Securities Australia Ltd Sell 1,750 (1,160) UBS Se	23-Apr-14	UBS Securities Australia Ltd	Sell	1,015	(116)	Ordinary
UBS Securities Australia Ltd Sell 3,229 (269) UBS Securities Australia Ltd Sell 30,721 (4,311) UBS Securities Australia Ltd Sell 3,722 (4,311) UBS Securities Australia Ltd Sell 3,623 (1,124) UBS Securities Australia Ltd Sell 3,633 (3,958) UBS Securities Australia Ltd Sell 10,223 (1,169) UBS Securities Australia Ltd Sell 10,220 (1,169) UBS Securities Australia Ltd Sell 1,420 (4,31) UBS Securities Australia Ltd Sell 1,420 (4,31) UBS Securities Australia Ltd Sell 1,210 (1,169) UBS Securities Australia Ltd Sell 1,311 (1,50) UBS Securities Australia Ltd Sell 1,310 (1,50) UBS Securities Australia Ltd Sell 1,326 (3,00) UBS Securities Australia Ltd Sell 1,300 (1,00) UBS Securities Australia Ltd Sell 1,300 (4,10) UBS	23-Apr-14	UBS Securities Australia Ltd	Seli	8,470	(896)	Ordinary
UBS Securities Australia Ltd Sell 35.05 (4.124) UBS Securities Australia Ltd Sell 37.721 (4.431) UBS Securities Australia Ltd Sell 3.623 (3.58) UBS Securities Australia Ltd Sell 3.623 (3.58) UBS Securities Australia Ltd Sell 1.64 (1.77) UBS Securities Australia Ltd Sell 1.66 (1.219) UBS Securities Australia Ltd Sell 1.64 (1.219) UBS Securities Australia Ltd Sell 1.64 (1.219) UBS Securities Australia Ltd Sell 1.20 (1.219) UBS Securities Australia Ltd Sell 1.21 (1.567) UBS Securities Australia Ltd Sell 1.22 (1.40) UBS Securities Australia Ltd Sell 1.22 (1.60) UBS Securities A	23-Apr-14	UBS Securities Australia Ltd	Sell	2,329	(392)	Ordinary
UBS Securities Australia Ltd Sell 37.721 (4.311) UBS Securities Australia Ltd Sell 3.529 (4.49) UBS Securities Australia Ltd Sell 1.5429 (4.49) UBS Securities Australia Ltd Sell 1.0229 (1.169) UBS Securities Australia Ltd Sell 1.0229 (1.169) UBS Securities Australia Ltd Sell 1.7420 (1.848) UBS Securities Australia Ltd Sell 1.7420 (1.848) UBS Securities Australia Ltd Sell 1.7420 (1.848) UBS Securities Australia Ltd Sell 1.750 (1.95) UBS Securities Australia Ltd Sell 1.750 (1.401) UBS Securities Australia Ltd Sell 1.750 (1.701) UBS Securities Australia Ltd Sell 1.750 (1.701) UBS Securities Australia Ltd Sell 1.750 (1.701) UBS Securities Australia Ltd Sell 1.750 (1.057) UBS Securities Australia Ltd Sell 1.750 (1.057)	23-Apr-14	UBS Securities Australia Ltd	Sell	36,085	(4,124)	Ordinary
UBS Securities Australia Ltd Sell 1,549 (449) UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 3,658 (1,169) UBS Securities Australia Ltd Sell 10,229 (1,169) UBS Securities Australia Ltd Sell 10,229 (1,169) UBS Securities Australia Ltd Sell 10,229 (1,169) UBS Securities Australia Ltd Sell 1,240 (1,19) UBS Securities Australia Ltd Sell 1,240 (1,219) UBS Securities Australia Ltd Sell 1,270 (449) UBS Securities Australia Ltd Sell 1,371 (1,557) UBS Securities Australia Ltd Sell 1,371 (1,557) UBS Securities Australia Ltd Sell 1,371 (1,557) UBS Securities Australia Ltd Sell 1,376 (1,57) UBS Securities Australia Ltd Sell 1,376 (1,57) UBS Securities Australia Ltd Sell 1,376 (1,57) UBS S	23-Apr-14	UBS Securities Australia Ltd	Sell	37 771	(4 311)	Ordinary
UBS Securities Australia Ltd Sell 1,529 (177) UBS Securities Australia Ltd Sell 34,633 (1,169) UBS Securities Australia Ltd Sell 10,229 (1,169) UBS Securities Australia Ltd Sell 10,229 (1,169) UBS Securities Australia Ltd Sell 1,420 (849) UBS Securities Australia Ltd Sell 1,420 (1,169) UBS Securities Australia Ltd Sell 1,213 (1,169) UBS Securities Australia Ltd Sell 1,213 (1,50) UBS Securities Australia Ltd Sell 1,233 (1,50) UBS Securities Australia Ltd Sell 1,235 (40) UBS Securities Australia Ltd Sell 1,235 (40) UBS Securities Australia Ltd Sell 1,205 (1,00) UBS Securities Australia Ltd Sell 1,206 (1,50) UBS Securities Australia Ltd Sell 1,206 (1,00) UBS Securities Australia Ltd Sell 1,506 (1,00) UBS Securit	23-Apr-14	UBS Securities Australia Ltd	3	3 070	(0//)	Coling
UBS Securities Australia Ltd Sell 34 633 (3,558) UBS Securities Australia Ltd Sell 13 (3) (4) UBS Securities Australia Ltd Sell 114 (1) (1) UBS Securities Australia Ltd Sell 7,270 (848) UBS Securities Australia Ltd Sell 1,371 (1) UBS Securities Australia Ltd Sell 1,370 (1) UBS Securities Australia Ltd Sell 1,370 (1) UBS Securities Australia Ltd Sell 1,330 (1) UBS Securities Australia Ltd Sell 1,330 (1) UBS Securities Australia Ltd Sell 1,300 (1) UBS Securities A	23-Apr-14	UBS Securities Australia I td	19	1 549	(771)	Ordinary
UBS Securities Australia Ltd Sell 1,259 UBS Securities Australia Ltd Sell 1,0229 (1,169) UBS Securities Australia Ltd Sell 2,6033 (1,169) UBS Securities Australia Ltd Sell 7,420 (848) UBS Securities Australia Ltd Sell 7,426 (150) UBS Securities Australia Ltd Sell 1,313 (150) UBS Securities Australia Ltd Sell 1,276 (150) UBS Securities Australia Ltd Sell 1,276 (150) UBS Securities Australia Ltd Sell 1,276 (160) UBS Securities Australia Ltd Sell 1,260 (170) UBS Securities Australia Ltd Sell 1,260 (170) UBS Securities Australia Ltd Sell 1,260 (170) UBS Securities Australia Ltd <td< td=""><td>23-Anr-14</td><td>I IRS Securities Australia I to</td><td> J</td><td>34622</td><td>(0.000)</td><td>Carling</td></td<>	23-Anr-14	I IRS Securities Australia I to	J	34622	(0.000)	Carling
UBS Securities Australia Ltd Sell 10,239 (1,169) UBS Securities Australia Ltd Sell 1,14 (13) UBS Securities Australia Ltd Sell 7,420 (848) UBS Securities Australia Ltd Sell 7,420 (848) UBS Securities Australia Ltd Sell 7,420 (848) UBS Securities Australia Ltd Sell 7,375 (150) UBS Securities Australia Ltd Sell 1,371 (1,507) UBS Securities Australia Ltd Sell 1,371 (1,507) UBS Securities Australia Ltd Sell 1,225 (140) UBS Securities Australia Ltd Sell 2,625 (300) UBS Securities Australia Ltd Sell 1,306 (1,69) UBS Securities Austral	23-Apr-14	I IRS Securities Australia I to	100	35	(0)	Coding
UBS Securities Australia Ltd Sell 1,0,223 (1,1,6) UBS Securities Australia Ltd Sell 1,14 (13) UBS Securities Australia Ltd Sell 1,20 (1,29) UBS Securities Australia Ltd Sell 1,29 (1,29) UBS Securities Australia Ltd Sell 1,313 (1,29) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 1,23 (1,40) UBS Securities Australia Ltd Sell 1,23 (1,40) UBS Securities Australia Ltd Sell 1,25 (1,40) UBS Securities Australia Ltd Sell 1,25 (1,40) UBS Securities Australia Ltd Sell 1,24 (1,57) UBS Securities Australia Ltd Sell 1,24 (1,10) UBS Securities Australia Ltd Sell 1,540 (1,10) UBS Securities Australia Ltd Sell 1,540 (1,10) UBS Securities Australia Ltd Sell 1,540 (1,05) UBS Securities Austral	23-Apr-14	I IRS Societies Australia 144	100	10,000	(4)	Crolling
UBS Securities Australia Ltd Sell 2,6533 (2,982) UBS Securities Australia Ltd Sell 1,420 (848) UBS Securities Australia Ltd Sell 1,420 (849) UBS Securities Australia Ltd Sell 1,656 (1,99) UBS Securities Australia Ltd Sell 2,048 (1,50) UBS Securities Australia Ltd Sell 2,048 (1,50) UBS Securities Australia Ltd Sell 1,273 (1,50) UBS Securities Australia Ltd Sell 1,275 (1,40) UBS Securities Australia Ltd Sell 1,256 (1,40) UBS Securities Australia Ltd Sell 1,706 (1,567) UBS Securities Australia Ltd Sell 1,706 (1,95) UBS Securities Australia Ltd Sell 1,500 (1,95) UBS Securiti	+1-1dV-02	Obs securities Australia Ltd	New York	10,229	(601,1)	Ordinary
UBS Securities Australia Ltd Sell 7,420 (848) UBS Securities Australia Ltd Sell 7,420 (848) UBS Securities Australia Ltd Sell 1,249 (1,219) UBS Securities Australia Ltd Sell 1,243 (1,219) UBS Securities Australia Ltd Sell 1,311 (1,567) UBS Securities Australia Ltd Sell 1,311 (1,567) UBS Securities Australia Ltd Sell 1,325 (1,400) UBS Securities Australia Ltd Sell 1,225 (1,400) UBS Securities Australia Ltd Sell 1,235 (1,400) UBS Securities Australia Ltd Sell 1,236 (1,500) UBS Securities Australia Ltd Sell 1,706 (1,150) UBS Securities Australia Ltd Sell 1,706 (1,150) UBS Securities Australia Ltd Sell 2,623 (4,130) UBS Securities Australia Ltd Sell 2,623 (3,00) UBS Securities Australia Ltd Sell 2,623 (4,130) UBS	23-Apr-14	UBS Securities Australia Ltd	Sell	26,093	(2,982)	Ordinary
UBS Securities Australia Ltd Sell 7,420 (848) UBS Securities Australia Ltd Sell 10,666 (1,219) UBS Securities Australia Ltd Sell 13,73 (1,219) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 1,313 (150) UBS Securities Australia Ltd Sell 1,373 (150) UBS Securities Australia Ltd Sell 1,275 (60) UBS Securities Australia Ltd Sell 1,276 (100) UBS Securities Australia Ltd Sell 1,276 (100) UBS Securities Australia Ltd Sell 1,276 (100) UBS Securities Australia Ltd Sell 1,390 (1,15) UBS Securities Australia Ltd Sell 1,500 (1,00) UBS Securities Australia Ltd Sell 1,500 (1,00) UBS Securities Australia Ltd Sell 1,500 (1,00) UBS Securities Australia Ltd Sell 2,625 (4,130) UBS Securities Aus	23-Apr-14	UBS Securities Australia Ltd	Sell	114	(13)	Ordinary
UBS Securities Australia Ltd Sell 10,666 (1,29) UBS Securities Australia Ltd Sell 375 (4) UBS Securities Australia Ltd Sell 4375 (500) UBS Securities Australia Ltd Sell 1,313 (1,527) UBS Securities Australia Ltd Sell 1,225 (140) UBS Securities Australia Ltd Sell 1,225 (140) UBS Securities Australia Ltd Sell 1,225 (140) UBS Securities Australia Ltd Sell 2,625 (300) UBS Securities Australia Ltd Sell 1,736 (140) UBS Securities Australia Ltd Sell 1,736 (140) UBS Securities Australia Ltd Sell 1,736 (1,95) UBS Securities Australia Ltd Sell 1,736 (1,95) UBS Securities Australia Ltd Sell 1,540 (1,150) UBS Securities Australia Ltd Sell 2,525 (300) UBS Securities Australia Ltd Sell 35,138 (4,130) UBS Securities Austral	23-Apr-14	UBS Securities Australia Ltd	Sell	7,420	(848)	Ordinary
UBS Securities Australia Ltd Sell 10,666 (1,219) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 2,048 (1,24) UBS Securities Australia Ltd Sell 1,313 (1,50) UBS Securities Australia Ltd Sell 1,275 (58) UBS Securities Australia Ltd Sell 1,275 (1,40) UBS Securities Australia Ltd Sell 1,255 (1,40) UBS Securities Australia Ltd Sell 1,260 (2,16) UBS Securities Australia Ltd Sell 1,890 (2,16) UBS Securities Australia Ltd Sell 1,890 (1,10) UBS Securities Australia Ltd Sell 1,706 (1,19) UBS Securities Australia Ltd Sell 3,625 (3,00) UBS Securities Australia Ltd Sell 3,50 (4,13) UBS Securities Australia Ltd Sell 3,50 (4,10) UBS Securities Australia Ltd Sell 2,625 (3,00) UBS Securities A	23-Apr-14	UBS Securities Australia Ltd	Sell	6	ε	Ordinary
UBS Securities Australia Ltd Sell 4.375 (4) UBS Securities Australia Ltd Sell 2,443 (500) UBS Securities Australia Ltd Sell 1,313 (1,567) UBS Securities Australia Ltd Sell 315 (588) UBS Securities Australia Ltd Sell 315 (580) UBS Securities Australia Ltd Sell 2,754 (100) UBS Securities Australia Ltd Sell 2,623 (100) UBS Securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 1,706 (1,907) UBS Securities Australia Ltd Sell 1,500 (1,907) UBS Securities Australia Ltd Sell 2,535 (1,007) UBS Securities Australia Ltd Sell 2,535 (1,007) UBS Securities Australia	23-Apr-14	UBS Securities Australia Ltd	Sell	10.666	(1 219)	Ordinary
UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 1,048 (234) UBS Securities Australia Ltd Sell 1,311 (1,567) UBS Securities Australia Ltd Sell 31,511 (1,567) UBS Securities Australia Ltd Sell 31,511 (1,567) UBS Securities Australia Ltd Sell 1,225 (140) UBS Securities Australia Ltd Sell 2,623 (300) UBS Securities Australia Ltd Sell 1,736 (1,130) UBS Securities Australia Ltd Sell 1,540 (1,156) UBS Securities Australia Ltd Sell 2,625 (3,012) UBS Securities Australia Ltd Sell 2,535 (4,00) UBS Securities Australia Ltd Sell 2,535 (1,00) UBS Se	23-Apr-14	UBS Securities Australia Ltd	Sell	35	(4)	Ordinary
UBS Securities Australia Ltd Sell 2,048 (234) UBS Securities Australia Ltd Sell 1,313 (150) UBS Securities Australia Ltd Sell 1,378 (152) UBS Securities Australia Ltd Sell 1,225 (100) UBS Securities Australia Ltd Sell 1,250 (216) UBS Securities Australia Ltd Sell 1,250 (216) UBS Securities Australia Ltd Sell 1,500 (175) UBS Securities Australia Ltd Sell 356,138 (4,130) UBS Securities Australia Ltd Sell 3,500 (1,097) UBS Securities Australia Ltd Sell 3,500 (1,097) UBS Securities Australia Ltd Sell 3,500 (4,00) UBS Securities Australia Ltd Sell 2,538 (4,00) UBS Securities Aus	23-Anr-14	II IRS Securities Australia I to	Call	375 /	(003)	Ordinan
UBS Securities Australia Ltd Sell 1,249 (1507) UBS Securities Australia Ltd Sell 1,313 (1,567) UBS Securities Australia Ltd Sell 1,225 (1400) UBS Securities Australia Ltd Sell 1,225 (1400) UBS Securities Australia Ltd Sell 2,625 (300) UBS Securities Australia Ltd Sell 2,625 (300) UBS Securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 35,138 (4,130) UBS Securities Australia Ltd Sell 35,53 (4,1) UBS Securities Australia Ltd Sell 35,53 (1,07) UBS Securities Australia Ltd Sell 35,53 (1,07) UBS Securities Australia Ltd Sell 35,50 (1,007) UBS Securities Australia Ltd Sell 3,375 (3,012) UBS Securities	23-Apr-14	I IBC Countition Australia 144	lles s	0,0,0	(000)	Oldinary
UBS Securities Australia Ltd Sell 1,513 (1,597) UBS Securities Australia Ltd Sell 375 (688) UBS Securities Australia Ltd Sell 1,225 (699) UBS Securities Australia Ltd Sell 7,954 (690) UBS Securities Australia Ltd Sell 22,033 (2,16) UBS Securities Australia Ltd Sell 2,625 (300) UBS Securities Australia Ltd Sell 1,390 (2,16) UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 35,138 (4,130) UBS Securities Australia Ltd Sell 35,61 (1,75) UBS Securities Australia Ltd Sell 35,63 (4,130) UBS Securities Australia Ltd Sell 6,17 (7,1) UBS Securities Australia Ltd Sell 2,535 (4,00) UBS Securities Australia Ltd Sell 2,538 (4,00) UBS Securiti	20 000	I DC Complete Australia Ltd	IIAS	2,040	(4.10)	Ordinary
UBS Securities Australia Ltd Sell 13.71 (1.567) UBS Securities Australia Ltd Sell 5.75 (6.90) UBS Securities Australia Ltd Sell 7.22 (14.0) UBS Securities Australia Ltd Sell 7.22 (9.09) UBS Securities Australia Ltd Sell 7.22 (9.09) UBS Securities Australia Ltd Sell 1.890 (2.518) UBS Securities Australia Ltd Sell 1.890 (1.15) UBS Securities Australia Ltd Sell 1.706 (1.15) UBS Securities Australia Ltd Sell 35.9 (4.1) UBS Securities Australia Ltd Sell 1.540 (1.06) UBS Securities Australia Ltd Sell 1.540 (1.07) UBS Securities Australia Ltd Sell 2.52 (3.00) UBS Securities Australia Ltd Sell 2.535 (4.00) UBS Securities Australia Ltd Sell 2.535 (4.00) UBS Securities Australia Ltd Sell 2.048 (5.00) UBS Securities Aus	23 - H	Opp Securities Australia Liu	Sell	(,313	(051)	Orginary
UBS Securities Australia Ltd Sell 3158 (658) UBS Securities Australia Ltd Sell 315 (36) UBS Securities Australia Ltd Sell 7,954 (909) UBS Securities Australia Ltd Sell 22,033 (300) UBS Securities Australia Ltd Sell 1,890 (216) UBS Securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 1,540 (176) UBS Securities Australia Ltd Sell 35,99 (4,130) UBS Securities Australia Ltd Sell 62,1 (7,1) UBS Securities Australia Ltd Sell 62,1 (7,1) UBS Securities Australia Ltd Sell 2,535 (4,00) UBS Securities Australia Ltd Sell 2,535 (4,00) UBS Securities Australia Ltd Sell 2,538 (1,007) UBS Securities Australia Lt	73-Apr-14	UBS Securities Australia Ltd	Sell	13,711	(1,567)	Ordinary
UBS Securities Australia Ltd Sell 1,225 (140) UBS Securities Australia Ltd Sell 7,954 (909) UBS Securities Australia Ltd Sell 22,033 (2,518) UBS Securities Australia Ltd Sell 2,625 (300) UBS Securities Australia Ltd Sell 1,330 (152) UBS Securities Australia Ltd Sell 36,138 (152) UBS Securities Australia Ltd Sell 36,138 (41) UBS Securities Australia Ltd Sell 35,535 (40) UBS Securities Australia Ltd Sell 31,063 (1,05) UBS Securities Australia Ltd Sell 62,1 (71) UBS Securities Australia Ltd Sell 62,1 (71) UBS Securities Australia Ltd Sell 62,1 (71) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 2,048 (2,54) UBS Securities Australia Ltd Sell 2,048 (2,54) UBS Securities Australia L	23-Apr-14	UBS Securities Australia Ltd	Sell	5,758	(658)	Ordinary
UBS Securities Australia Ltd Sell 1,225 (140) UBS Securities Australia Ltd Sell 2,625 (309) UBS Securities Australia Ltd Sell 2,625 (300) UBS Securities Australia Ltd Sell 1,890 (216) UBS Securities Australia Ltd Sell 1,706 (152) UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 35,93 (4,130) UBS Securities Australia Ltd Sell 31,663 (4,130) UBS Securities Australia Ltd Sell 31,663 (4,00) UBS Securities Australia Ltd Sell 62,1 (7) UBS Securities Australia Ltd Sell 6,235 (4,00) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 2,048 (4,00) UBS Securities Australia Ltd Sell 2,048 (2,40) UBS Securities	23-Apr-14	UBS Securities Australia Ltd	Sell	315	(36)	Ordinary
UBS securities Australia Ltd 5ell 7,954 (999) UBS Securities Australia Ltd 5ell 2,5033 (2,518) UBS Securities Australia Ltd 5ell 1,890 (2,16) UBS Securities Australia Ltd 5ell 1,330 (1,52) UBS Securities Australia Ltd 5ell 1,706 (195) UBS Securities Australia Ltd 5ell 3,525 (4,130) UBS Securities Australia Ltd 5ell 3,525 (4,130) UBS Securities Australia Ltd 5ell 3,525 (4,130) UBS Securities Australia Ltd 5ell 6,21 (7,1) UBS Securities Australia Ltd 5ell 6,21 (7,1) UBS Securities Australia Ltd 5ell 6,21 (7,1) UBS Securities Australia Ltd 5ell 2,535 (4,00) UBS Securities Australia Ltd 5ell 2,535 (4,00) UBS Securities Australia Ltd 5ell 2,538 (4,00) UBS Securities Australia Ltd 5ell 2,548 (4,00) UBS Securities A	23-Apr-14	UBS Securities Australia Ltd	Sell	1 225	(140)	Ordinary
UBS Securities Australia Ltd Sell 22,033 (2,538) UBS Securities Australia Ltd Sell 1,890 (2,538) UBS Securities Australia Ltd Sell 1,890 (1,65) UBS Securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 35,638 (41) UBS Securities Australia Ltd Sell 1,540 (17) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 6,375 (500) UBS Securities Australia Ltd Sell 2,375 (500) UBS Securities Australia Ltd Sell 2,388 (400) UBS Securities Australia Ltd Sell 2,388 (400) UBS Securities Australia Ltd Sell 2,048 (234) UBS Securities Australia Ltd<	23-Apr-14	UBS Securities Australia I to	Sell	7 957	(000)	Ordinary
USS Securities Australia Ltd Sell 2,625 (2.0) USS Securities Australia Ltd Sell 2,625 (3.00) USS Securities Australia Ltd Sell 1,330 (1.52)	23 Apr 14	1 DC Countition Australia 144	100	200,00	(606)	Ciciliaiy
UBS Securities Australia Ltd Sell 1,525 (300) UBS Securities Australia Ltd Sell 1,330 (152) UBS Securities Australia Ltd Sell 1,330 (152) UBS Securities Australia Ltd Sell 35,136 (153) UBS Securities Australia Ltd Sell 35,535 (41) UBS Securities Australia Ltd Sell 31,663 (41) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 623 (1,097) UBS Securities Australia Ltd Sell 623 (1,097) UBS Securities Australia Ltd Sell 2,535 (400) UBS Securities Australia Ltd Sell 2,000 (400) UBS Securities Australia Ltd Sell 2,048 (2,34) UBS Securities Australia Ltd Sell 2,048 (2,34) UBS Securities Australia Ltd	23-Api-14	Obsideration Australia Ltd	Sell:	22,033	(815,2)	Ordinary
UBS Securities Australia Ltd Sell 1,890 (216) UBS Securities Australia Ltd Sell 1,330 (152) UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 2,625 (4,130) UBS Securities Australia Ltd Sell 35,9 (4,130) UBS Securities Australia Ltd Sell 35,9 (4,130) UBS Securities Australia Ltd Sell 62,1 (7) UBS Securities Australia Ltd Sell 62,1 (7) UBS Securities Australia Ltd Sell 6,375 (5,00) UBS Securities Australia Ltd Sell 26,355 (1,007) UBS Securities Australia Ltd Sell 26,355 (4,007) UBS Securities Australia Ltd Sell 2,048 (2,000) UBS Securities Australia Ltd Sell 2,048 (2,000) UBS Securities Australia Ltd Sell 2,048 (400) UBS Securities Australia Ltd Sell 2,048 (424) UBS Securities A	73-Apr-14	UBS Securities Australia Ltd	Sell	2,625	(300)	Ordinary
UBS Securities Australia Ltd Sell 1,330 (122) UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 2,625 (300) UBS Securities Australia Ltd Sell 1,540 (1,130) UBS Securities Australia Ltd Sell 1,540 (1,15) UBS Securities Australia Ltd Sell 62,1 (7,1) UBS Securities Australia Ltd Sell 62,1 (7,1) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 4,375 (3,002) UBS Securities Australia Ltd Sell 29,599 (1,097) UBS Securities Australia Ltd Sell 29,599 (1,097) UBS Securities Australia Ltd Sell 29,383 (3,002) UBS Securities Australia Ltd Sell 29,383 (4,00) UBS Securities Australia Ltd Sell 2,048 (2,34) UBS Securities Australia Ltd Sell 2,048 (3,34) UBS Secur	23-Apr-14	UBS Securities Australia Ltd	Sell	1,890	(216)	Ordinary
UBS securities Australia Ltd Sell 1,706 (195) UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 35,9 (41) UBS Securities Australia Ltd Sell 1,540 (176) UBS Securities Australia Ltd Sell 62,1 (71) UBS Securities Australia Ltd Sell 62,1 (71) UBS Securities Australia Ltd Sell 9,599 (1,007) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 2,355 (3,012) UBS Securities Australia Ltd Sell 2,355 (400) UBS Securities Australia Ltd Sell 2,358 (400) UBS Securities Australia Ltd Sell 2,048 (2,37) UBS Securities Australia Ltd Sell 2,048 (2,49) UBS Securities Australia Ltd Sell 2,048 (424) UBS Securities Australia Ltd Sell 2,048 (424) UBS Securities Australia	23-Apr-14	UBS Securities Australia Ltd	Sell	1,330	(152)	Ordinary
UBS Securities Australia Ltd Sell 36,138 (4,130) UBS Securities Australia Ltd Sell 3.59 (41) UBS Securities Australia Ltd Sell 1,540 (176) UBS Securities Australia Ltd Sell 1,540 (176) UBS Securities Australia Ltd Sell 62.1 (71) UBS Securities Australia Ltd Sell 62.1 (71) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 2,535 (1,097) UBS Securities Australia Ltd Sell 2,635 (400) UBS Securities Australia Ltd Sell 2,635 (400) UBS Securities Australia Ltd Sell 2,938 (400) UBS Securities Australia Ltd Sell 2,948 (400) UBS Securities Australia Ltd Sell 2,048 (449) UBS Securities Australia Ltd Sell 2,048 (449) UBS Securities Australia Ltd Sell 2,048 (449) UBS Securities Australia Ltd<	23-Apr-14	UBS Securities Australia Ltd	Sell	1,706	(195)	Ordinary
UBS Securities Australia Ltd Sell 2,625 (41) UBS Securities Australia Ltd Sell 1,540 (41) UBS Securities Australia Ltd Sell 1,540 (176) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 9,599 (1,097) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 2,535 (3,012) UBS Securities Australia Ltd Sell 2,938 (3,580) UBS Securities Australia Ltd Sell 2,048 (2,34) UBS Securities Australia Ltd Sell 3,029 (449) UBS Securities Australia	23-Apr-14	UBS Securities Australia Ltd	ll d'S	36 138	(4 130)	Ordinary
UBS Securities Australia Ltd Sell 55.25 (4.0) UBS Securities Australia Ltd Sell 1.540 (1/6) UBS Securities Australia Ltd Sell 62.1 (7.1) UBS Securities Australia Ltd Sell 62.1 (7.1) UBS Securities Australia Ltd Sell 62.1 (7.1) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 26,355 (3,012) UBS Securities Australia Ltd Sell 26,355 (4,007) UBS Securities Australia Ltd Sell 26,355 (4,007) UBS Securities Australia Ltd Sell 2,048 (2,000) UBS Securities Australia Ltd Sell 2,048 (2,000) UBS Securities Australia Ltd Sell 2,048 (249) UBS Securities Australia Ltd Sell 2,048 (249) UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 3,539 (409) UBS Securities Aus	23.Anr.14	I IRC Securities Australia 14d	= 0	3676	(000)	Ordinan,
UBS Securities Australia Ltd Sell 1,540 (176) UBS Securities Australia Ltd Sell 3,165 (1,550) UBS Securities Australia Ltd Sell 62.1 (71) UBS Securities Australia Ltd Sell 62.1 (71) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 7,500 (2,000) UBS Securities Australia Ltd Sell 7,500 (2,000) UBS Securities Australia Ltd Sell 7,500 (2,000) UBS Securities Australia Ltd Sell 2,048 (3,358) UBS Securities Australia Ltd Sell 2,048 (2,334) UBS Securities Australia Ltd Sell 2,048 (2,34) UBS Securities Australia Ltd Sell 3,529 (419) UBS Securities A	23-Anr-14	I IRS Securities Australia I td		010,5	(141)	Calibro
UBS Securities Australia Ltd Sell 1,740 UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 9,599 (1,097) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 7,500 (2,000) UBS Securities Australia Ltd Sell 28,761 (400) UBS Securities Australia Ltd Sell 2,048 (2,34) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 3,539 (4,29) UBS Securities Australia Ltd Sell 3,539 (4,29) UBS Securities Australia Ltd	25 Apr 1.4	Let Charter A reference		1 140	()(1)	9
UBS Securities Australia Ltd Sell 6.17.05 UBS Securities Australia Ltd Sell 6.21 (7.1) UBS Securities Australia Ltd Sell 6.21 (7.1) UBS Securities Australia Ltd Sell 4.375 (5.00) UBS Securities Australia Ltd Sell 7.500 (5.00) UBS Securities Australia Ltd Sell 7.500 (2.000) UBS Securities Australia Ltd Sell 2.53.86 (4.00) UBS Securities Australia Ltd Sell 2.54.61 (4.00) UBS Securities Australia Ltd Sell 2.54.61 (4.20) UBS Securities Australia Ltd Sell 2.048 (2.54) UBS Securities Australia Ltd Sell 2.048 (2.49) UBS Securities Australia Ltd Sell 2.048 (2.49) UBS Securities Australia Ltd Sell 2.048 (2.49) UBS Securities Australia Ltd Sell 2.5339 (4.69) UBS Securities Australia Ltd Sell 2.5339 (4.69) UBS Securities Australia Ltd </td <td>100.00</td> <td>ODS Securities Australia Ltd</td> <td> " C</td> <td>046,1</td> <td>(0/1)</td> <td>Ordinary</td>	100.00	ODS Securities Australia Ltd	" C	046,1	(0/1)	Ordinary
UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 621 (71) UBS Securities Australia Ltd Sell 9,599 (1,097) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 17,500 (2,002) UBS Securities Australia Ltd Sell 29,383 (3,358) UBS Securities Australia Ltd Sell 28,761 (400) UBS Securities Australia Ltd Sell 2,048 (2,334) UBS Securities Australia Ltd Sell 3,520 (449) UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 3,539 (4,19) UBS Securities Aust	73-Apr-14	UBS Securities Australia Ltd	Sell	31,063	(3,550)	Ordinary
UBS Securities Australia Ltd Sell (71) UBS Securities Australia Ltd Sell 9.599 (1,097) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 17,500 (2,000) UBS Securities Australia Ltd Sell 25,388 (4,000) UBS Securities Australia Ltd Sell 25,661 (400) UBS Securities Australia Ltd Sell 2,048 (240) UBS Securities Australia Ltd Sell 2,048 (244) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 3,523 (449) UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 3,539 (4,13) UBS Securities Australia Ltd Sell 3,539 (4,13) UBS Securities Australia Ltd Sell 3,539 (4,149) UBS Securities Australia Ltd	23-Apr-14	UBS Securities Australia Ltd	Sell	621	(11)	Ordinary
UBS Securities Australia Ltd Sell 9,599 (1,097) UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 7,500 (2,000) UBS Securities Australia Ltd Sell 7,500 (2,000) UBS Securities Australia Ltd Sell 7,500 (2,000) UBS Securities Australia Ltd Sell 2,938 (3,389) UBS Securities Australia Ltd Sell 2,048 (2,387) UBS Securities Australia Ltd Sell 2,048 (2,387) UBS Securities Australia Ltd Sell 2,048 (2,387) UBS Securities Australia Ltd Sell 2,048 (2,531) UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 3,539 (4,10) UBS Securities Australia Ltd Sell 3,539 (4,10) UBS Securities Australia Ltd Sell 3,539 (4,10)	23-Apr-14	UBS Securities Australia Ltd	Sell	621	(71)	Ordinary
UBS Securities Australia Ltd Sell 4,375 (500) UBS Securities Australia Ltd Sell 7,575 (3,012) UBS Securities Australia Ltd Sell 17,500 (2,000) UBS Securities Australia Ltd Sell 23,833 (4,00) UBS Securities Australia Ltd Sell 28,761 (400) UBS Securities Australia Ltd Sell 2,048 (2,347) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 3,539 (4,13) UBS Securities Australia Ltd Sell 3,529 (4,49) UBS Securities Australia Ltd Sell 3,539 (4,19) UBS Securities Australia Ltd Sell 3,539 (4,19)	23-Apr-14	UBS Securities Australia Ltd	Sell	665'6	(1,097)	Ordinary
UBS Securities Australia Ltd Sell 75,555 (3,02) UBS Securities Australia Ltd Sell 17,500 (2,000) UBS Securities Australia Ltd Sell 3,500 (400) UBS Securities Australia Ltd Sell 2,046 (2,34) UBS Securities Australia Ltd Sell 2,046 (2,34) UBS Securities Australia Ltd Sell 2,046 (449) UBS Securities Australia Ltd Sell 2,339 (449) UBS Securities Australia Ltd Sell 2,339 (4,13) UBS Securities Australia Ltd Sell 3,539 (4,13) UBS Securities Australia Ltd Sell 3,539 (4,13) UBS Securities Australia Ltd Sell 3,539 (4,13)	23-Apr-14	UBS Securities Australia Ltd	Sett	4,375	(200)	Ordinary
UBS Securities Australia Ltd Sell 2,0,50 1,0,00 UBS Securities Australia Ltd Sell 7,938 (3,388) UBS Securities Australia Ltd Sell 29,383 (3,388) UBS Securities Australia Ltd Sell 28,761 (400) UBS Securities Australia Ltd Sell 2,048 (2,387) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 3,529 (419) UBS Securities Australia Ltd Sell 3,529 (410) UBS Securities Australia Ltd Sell 8,578 (1,026)	23-Anr-14	I IBS Securities Australia 14d	160	26.36	(0.00)	Compa
USS Securities Australia Ltd Sell 7,500 UBS Securities Australia Ltd Sell 3,500 (400) UBS Securities Australia Ltd Sell 3,500 (400) UBS Securities Australia Ltd Sell 3,500 (400) UBS Securities Australia Ltd Sell 2,048 (249) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 3,539 (4,15) UBS Securities Australia Ltd Sell 3,529 (4,15) UBS Securities Australia Ltd Sell 3,539 (4,10)	22 Apr 14	1 DC Committee Australia Ltd	100	20,23	(210,0)	S C
USS Securities Australia Ltd Sell 2,5380 UBS Securities Australia Ltd Sell 2,048 UBS Securities Australia Ltd Sell 2,048 UBS Securities Australia Ltd Sell 2,048 UBS Securities Australia Ltd Sell 2,5339 UBS Securities Australia Ltd Sell 2,5339 UBS Securities Australia Ltd Sell 2,5339 UBS Securities Australia Ltd Sell 3599 (4,1) UBS Securities Australia Ltd Sell 8,978 (1,026)	20 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -	Opposed interpretation	i o	006,11	(2,000)	Ordinary
UBS Securities Australia Ltd Sell 3,500 (400) UBS Securities Australia Ltd Sell 28,761 (3,237) UBS Securities Australia Ltd Sell 2,048 (2,34) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 3,529 (41) UBS Securities Australia Ltd Sell 8,978 (1,026)	43-Api-14	Obs securities Australia Ltd	Sell	29,383	(3,338)	Ordinary
UBS Securities Australia Ltd Sell 28,761 (3,287) UBS Securities Australia Ltd Sell 2,048 (234) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 22,339 (2,533) UBS Securities Australia Ltd Sell 35,929 (41) UBS Securities Australia Ltd Sell 35,929 (41) UBS Securities Australia Ltd Sell 35,929 (41)	23-Apr-14	UBS Securities Australia Ltd	Sell	3,500	(400)	Ordinary
UBS Securities Australia Ltd Sell 2,048 (234) UBS Securities Australia Ltd Sell 3,929 (449) UBS Securities Australia Ltd Sell 22,339 (2,533) UBS Securities Australia Ltd Sell 359 (1,026)	23-Apr-14	UBS Securities Australia Ltd	Sell	28,761	(3,287)	Ordinary
UBS Securities Australia Ltd Sell 3,529 (449) UBS Securities Australia Ltd Sell 22,339 (2,533) UBS Securities Australia Ltd Sell 359 (41) UBS Securities Australia Ltd Sell 8,978 (1,026)	23-Apr-14	UBS Securities Australia Ltd	Sell	2,048	(234)	Ordinary
UBS Securities Australia Ltd Sell 22,339 (2,553) UBS Securities Australia Ltd Sell 359 (41) UBS Securities Australia Ltd Sell 8,978 (1,026)	23-Apr-14	UBS Securities Australia Ltd	Sell	3,929	(449)	Ordinary
UBS Securities Australia Ltd Sell 359 (41) UBS Securities Australia Ltd Sell 8,978 (1,026)	23-Apr-14	UBS Securities Australia Ltd	Sell	22,339	(2,553)	Ordinary
UBS Securities Australia Ltd Sell 8,978 (1,026)	23-Apr-14	UBS Securities Australia Ltd	Sell	329	(41)	Ordinary
(370') Securities Australia Ltd Securities Australia Ltd Securities Australia Ltd	23. Apr. 14	I IDC Countition Australia 14d	100	0 020	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
The second secon	40-MI-14	UBS Securities Australia Liu	- Carrier	- ()		Overling

22-Apr-14 22-Apr-14 22-Apr-14	USC Securities Australia Ltd	Buy	11 500	104/41	Ordinary
Apr-14		200		1 3355	Ordina.
-Apr-14	URS Securities Australia 14d	N. B	737 175	31 696	
	IRS Securities Australia 1td	200	767,27 367,36	(170.5)	
2 100 1	De Courties Australia Liu	100	960 (64.4	(1,0,0)	
2 Apr 14	Pal ciletter & continues con	Party Designation	4,422,930	2/0,016	Ordinary
2 Apr. 14	UDC Cocurition Australia Ltd	Enter Into Derivative	V. V.	(001)	
2-Anr-14	URS Wealth Management Australia (14	Buy	1 005	361	Graining
2-Anr-14	IRS Wealth Management Australia Ltd	200	9631	176	
22.00-14	1 IDC 1//colth Macacoccoch Australia 14d	, , ,	0000		
11	Service Administration of the service Authority Services	Ana a	040,0	002	Ordinary
2 Apr 14	USS Seatt I Mai lagel left Australia Ltd	fina 3	024 00		
11-104	Obs Securities Australia Lid	Jan J	0/4/67		
22-Api-14	UBS Securities Australia Ltd	Sell	42,243	(4,8/8)	
#1-Idw	UBS Securities Australia Lid	36	000,00	(4,429)	
22-Apr-14	UBS Securities Australia Ltd	8	90,679	(10,471)	Ordinary
Apr-14	UBS Securities Australia Ltd	ll Ne	/,846	(906)	Ordinary
2-Apr-14	UBS Securities Australia Ltd	lia.	761,08	(3,487)	
Apr-14	UBS Securities Australia Ltd	Se.	28,162	(3,252)	
2-Apr-14	UBS Securities Australia Ltd	Sei	45,448	(5,248)	ŀ
2-Apr-14	UBS Securities Australia Ltd	Ş.	17,277	(1,995)	Ordinary
2-Apr-14	UBS Securities Australia Ltd	Sell	1,143	(132)	
-Apr-14	UBS Securities Australia Ltd	lles.	3,464	(400)	
-Apr-14	UBS Securities Australia Ltd	Seli	9,405	(1,086)	
22-Apr-14	UBS Securities Australia Ltd	Sell	6,353	(1,080)	
2-Apr-14	UBS Securities Australia Ltd	Sell	1,931	(223)	
2-Apr-14	UBS Securities Australia Ltd	Sell	61,460	(7,097)	Ordinary
2-Apr-14	UBS Securities Australia Ltd	Sell	1,593	(184)	
Apr-14	UBS Securities Australia Ltd	Stock returned	N/A	(57,309)	
3-Apr-14	UBS Securities Australia Ltd	Buy	2,620	300	L
3-Apr-14	UBS Securities Australia Ltd	Seil	609'26	(11,150)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Buy	70,480	880'8	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	174,616	(20,000)	
23-Apr-14	UBS Securities Australia Ltd	Sell	21,850	(2,500)	Ordinary
Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(523)	
3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(1,000)	Options
3-Apr-14	UBS Securities Australia Ltd	Enter into Derivative (NA	(810)	
3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	225	Options
3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(2)	Options
3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative Q	N/A	300	Options
3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (ΝA	216	
3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	(2,750)	
3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	(1,121)	
3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(249)	
-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(200)	Options
23-Apr-14	UBS Global Asset Management	Sell	363,649	(41,884)	
3-Apr-14	UBS Global Asset Management	Buy	38,915	4,446	Ordinary
3-Apr-14	UBS Global Asset Management	Buy	3,317	379	Ordinary
3-Apr-14	UBS Wealth Management Australia Ltd	lleS.	3,000	(343)	
3-Apr-14	UBS Securities Australia Ltd	lles.	18,541	(2,119)	
3-Apr-14	UBS Securities Australia Ltd	Sell	4,393	(205)	
3-Apr-14	UBS Securities Australia Ltd	les	1,943	(222)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	lis.	10,955	(1,252)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	l es	8.698	(964)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	S	131	(15)	Ordinary
3-Apr-14	t BS Societies Australia 11d	i i	169	(17)	Ordinan
7	1 BC Countries Australia 144	8 5	100	(46)	Ordinary
3 Apr. 14	USO Securities Australia Ltd	100	966	(96)	Ordinary
3-Anr-14	IRS Securities Australia I to	las	131	(15)	Ordinary
3-Anr-14	I BS Securities Australia I to	Sell	E7E 1E	(3 582)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	6	(1)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	929	(75)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	2,529	(289)	Ordinary
	UBS Securities Australia Ltd	Sell	4,104	(469)	Ordinary
١,	UBS Securities Australia Ltd	Sell	481	(22)	Ordinary
ı	UBS Securities Australia Ltd	Sell	1,715	(961)	Ordinary
ı	UBS Securities Australia Ltd	Sell	1,225	(140)	Ordinary
Г	UBS Securities Australia Ltd	Sell	3,684	(421)	Ordinary

URS Securities Australia Ltd Sell URS	١	מספ מספת וונים ויומים בומ		10.10	(nnc)	
UBS Securities Australia Ltd Sell UBS Se	- 1	UBS Securities Australia Ltd	Sell	2,087	(238)	Ordinary
UBS Securities Australia Ltd Sell UBS Se	- 1	UBS Securities Australia Ltd	Seil	87,700	(10,000)	Ordinary
UBS Securities Australia Ltd Sell UBS	- 1	UBS Securities Australia Ltd	Sell	277,7	(988)	Ordinary
UBS Securities Australia Ltd Sell UBS	-Apr-14	UBS Securities Australia Ltd	Sell	16,287	(1,855)	Ordinary
UBS Securities Australia Ltd Sell UBS	Apr-14	UBS Securities Australia Ltd	Sell	6,581	(052)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell Securities Australia Ltd Sell UBS Securi	8-Apr-14	UBS Securities Australia Ltd	Sell	2,027	(231)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	Apr-14	UBS Securities Australia Ltd	Sell	8,775	(1,000)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	1-Apr-14	UBS Securities Australia Ltd	Sell	8,775	(1,000)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	1-Apr-14	UBS Securities Australia Ltd	Self	8,775	(1,000)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	Apr-14	UBS Securities Australia Ltd	Sell	8,775	(1,000)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	Apr-14	UBS Securities Australia Ltd	Sell	8,775	(1,000)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	Apr-14	UBS Securities Australia Ltd	Sell	7,117	(811)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	23-Apr-14	UBS Securities Australia Ltd	Sell	3.412	(389)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	1-Apr-14	UBS Securities Australia Ltd	Setl	1,904	(217)	Ordinary
UBS Securities Australia Ltd UBS Securities A	-Apr-14	UBS Securities Australia Ltd	Spill	1 220	(140)	College
UBS Securities Australia Ltd UBS Securities A	-Apr-14	UBS Securities Australia Ltd	las	2.053	(DEC)	Ordinary
UBS Securities Australia Ltd UBS Securities A	-Apr-14	UBS Securities Australia Ltd	Sell	8 775	(1,000)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	Apr-14	UBS Securities Australia Ltd	Sell	623	(71)	Ordinary
UBS Securities Australia Ltd Sell UBS Se	-Apr-14	UBS Securities Australia Ltd	Sell	1609	(7.1)	Ordinary
UBS Securities Australia Ltd Sell (185 Securities Australia Ltd (185 Securities Australia L	-Apr-14	UBS Securities Australia Ltd	Sell	603	(71)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral		UBS Securities Australia Ltd	Sell	8.565	(475)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	1	UBS Securities Australia Ltd	Sell	746	(85)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd UBS Securities Australia Ltd Sell UBS Securities Australia Ltd UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd UBS Securities Australia Ltd Sell UBS Securities Australia Ltd UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia L	1	UBS Securities Australia Ltd	Seil	5 013	(571)	Ordinan
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia	1	UBS Securities Australia Ltd	Sell	5 268	(1 (6)	Ordinary
UBS Securities Australia Ltd Sell (UBS Securities Australia Ltd (UBS Securities Austra	-Apr-14	UBS Securities Australia Ltd	lles	115 756	(13 184)	Ording
UBS Securities Australia Ltd Sell UBS Securities Austral	-Apr-14	UBS Securities Australia Ltd	Sell	5.268	(600)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd UBS Securities Australia Ltd Sell UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia L	Apr-14	UBS Securities Australia Ltd	Sell	61.434	(1,007)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	-Apr-14	UBS Securities Australia Ltd	lle S	6779	(092)	Coding
UBS Securities Australia Ltd Sell UBS Se	-Apr-14	UBS Securities Australia Ltd	3	113 969	(13.025)	Graining
UBS Securities Australia Ltd Sell UBS Securities Austral	-Apr-14	UBS Securities Australia I td	- a	089	(620,01)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	-Apr-14	LIBS Securities Australia Ltd	3	1 000	(210)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	-Apr-14	UBS Securities Australia Ltd	-	103	(717)	Cidinary
UBS Securities Australia Ltd Sell UBS Securities Austral	Apr-14	UBS Securities Australia Ltd	Sell	2 354	(090)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral	Apr-14	UBS Securities Australia Itd	<u></u>	5,460	(625)	ording.
UBS Securities Australia Ltd Sell UBS Securities Austral	-Apr-14	UBS Securities Australia Ltd	S	1,006	(115)	Ordinary
UBS Securities Australia Ltd Seil UBS Securities Austral	-Apr-14	UBS Securities Australia Ltd	S	3005	(3/8)	Ording
UBS Securities Australia Ltd Sell UBS Securities Austral	-Apr-14	UBS Securities Australia 11d	9	621	(17)	Ordinary Strains
UBS Securities Australia Ltd Sell UBS Securities Austral	1	I IBS Securities Australia 14d	3	020	(1/)	Oromary
UBS Securities Australia Ltd Sell UBS Securities Austral	1	LIBS Securities Australia 11d	9	0,010	(1041)	Ordingry
UBS Securities Australia Ltd Sell UBS Securities Austral	1	LIRS Sequitities Australia 14d	190	901,6	(1,041)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	1	UBS Securities Australia I td	la	11,655	(1 332)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Austral		LIBS Securities Australia 11d	9	100	(17)	Grainary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	I IRS Securities Australia 1td	Col	170	(71)	Orginary
USS Securities Australia Ltd Sell UBS	700	Tipe Countries Australia Lite	II-S	170	(17)	Ordinary
USS Securities Australia Ltd Sell	1 200	Ups Securities Australia Etc) oei	14,595	(899,1)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	4D-14	Ubs securities Australia Ltd	Nell Nell	1,584	(181)	Ordinary
UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli UBS Securities Australia It do Seli	Apr-14	UBS Securities Australia Ltd	Sell	621	(71)	Ordinary
UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell Sell UBS Securities Australia Ltd Sell Sell Sell Sell Sell Sell Securities Australia Ltd Sell Sell Sell Sell Securities Australia Ltd Sell Sell Sell Sell Securities Australia Ltd Sell Sell Sell Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Sell Sell Sell Securities Australia Ltd Sell Sell Sell Securities Australia Ltd Sell	Apr-14	UBS Securities Australia Ltd	Seli	2,301	(263)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd	-Apr-14	UBS Securities Australia Ltd	Sell	621	(71)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	32,506	(3,715)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	16,266	(1,859)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	621	(71)	Ordinary
UBS securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	648	(74)	Ordinary
UBS securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	14,630	(1,672)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	630	(72)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	6.939	(262)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	Apr-14	UBS Securities Australia Ltd	Sell	689	(73)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	2,126	(243)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	15,278	(1,746)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	-Apr-14	UBS Securities Australia Ltd	Sell	621	(12)	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	Apr-14	UBS Securities Australia Ltd	II-9K	1.566	(179)	Ordinary
UBS Securities Australia Lid Sell UBS Securities Australia Ltd Sell	Apr-14	UBS Securities Australia Ltd	Sel	089	(27)	Ordina
UBS Securities Australia Ltd Sell	Apr-14	UBS Securities Australia Ltd	100	18 373	(7 004)	Ordinary
	Apr-14	UBS Securities Australia Itd		621	(12)	Ordinary
Call	Apr-14	I IRS Securities Australia Ad	S College	12021	(1/1)	Ordinary Ordinary
100 Company of the part of	1/	Let Chamber A solding to the	100	100,0	(6/6)	Coloniary
3-Apr-14 UBS Securities Australa Ltd Sell	Apr-14	UBS Securities Australia Ltd	Ne:	3,535	(404)	Ordinary
LODS SECULIES AUSTRIA LIG	ANT. 14	II IDC Con writing Alletralia to	-	0.000	(3011)	

URS Securities Australia Ltd Seaf 2,153 (246) URS Securities Australia Ltd Seal 2,703 (246) URS Securities Australia Ltd Seal 2,703 (246) URS Securities Australia Ltd Seal 4,900 (500) URS Securities Australia Ltd Seal 1,717 (247) URS Securities Australia Ltd Seal 2,723 (246) URS Securities Australia Ltd Seal 2,723 (246) URS Securities Australia Ltd Seal 2,723 (247) URS Securities Australia Ltd Seal 2,723 (246) URS Securities Australia Ltd Seal 1,715 (246) (1,717) URS Securities Australia Ltd Seal 1,716 (1,717) (1,717) <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>Ordinary</th> <th>, median</th> <th>Ordinary</th>	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	, median	Ordinary
Sell Sell	(36)	(246)	(303)	(005)	(094)	(022)	(6/7)	(212)	(254)	(36)	(750)	(177)	(232)	(387)	(449)	(69)	(63)	(504)	(622)	(254)	(772)	(7,292)	(11)	(1,162)	(163)	(449)	(122)	(1,000)	(1,308)	(1,000)	(2,435)	(217)	(36)	(4,582)	(211)	(779)	(96)	(1,500)	(748)	(487)	(989)	(151)	(360)	(10,543)	(199)	(446)	(1,500)	(6/7)	(157)	(3,289)	(1,500)	(138)	(70)	(415)	1000	(301)
	315	2,133	4,705	4,375	4,900	2000	0,010	21 875	2,273	315	6,563	6,746	2,057	3,386	3,929	516	551	8.750	6,816	2,223	6,755	11,305	96	10,168	1,426	3,929	1,068	8 750	11,445	8,750	21,306	1,904	316	40,184	1,850	6,832	316	13,163	6,560	4,271	6,016	4.841	3,157	92,462	1,745	3,911	13,163	6,832	1 333	28,845	13,163	1,210	614	3,640	0 212	8,213
	Self	Sell	Jes	llac Lac	Sell	ii o	lies.	i ay	iles.	les	Sell	Sell	Sell	Seil	Set	Sell	Sell	N 3	Sell	Sell	Sell	Sell	lles:	Sell	Sell	Sell	Sell	New York	lles	Sell	Sell	Sell	Sel	Sell	Sell	Sell	100	Sell	Sell	Sell	Sell	Sell	Sell	Seil	Sell	les!	les.	Sell		Sell	Sell	Sell	Sell	Seli		Sell
	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Lid	Securities Australia 14d	Societies Australia Ind	Securities Australia I fd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia I td	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd			Securities Australia Ltd	Securities Australia Ltd			Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	or cherry of the Contract of t	

23-Apr-14 UBS Securities Australia Ltd 23-Apr-14 UBS Sec			656 656 2,625 5,906 3,150	(75) (75) (300) (675) (675) (360) (234)	Ordinary Ordinary Ordinary Ordinary
			656 2,625 5,906 3,150	(75) (300) (675) (360) (234) (150)	Ordinary Ordinary Ordinary
			2,625 5,906 3,150	(300) (675) (360) (234) (150)	Ordinary Ordinary
			3,150	(675) (360) (234) (150)	Ordinary
		33, 33, 33, 33, 33, 33, 33, 33, 33, 33,	3,150	(34)	Ordinary
			0700	(150)	, ;;;;
			1.040.7	(150)	Ordinary
			1,313		Ordinary
		Sell Sell Sell	1,689	(193)	Ordinary
		Sell	2 573	(1007)	Ordinan
		Sell	1 803	(206)	oregip (
		Sell	136	(502)	California
		- Sei	10/	(8/)	Ordinary
			13,125	(1,500)	Ordinary
		Jak I	7,901	(803)	Ordinary
		Sell	28,779	(3,289)	Ordinary
		Sell	990'5	(625)	Ordinary
		Sell	8,750	(1,000)	Ordinary
		Sell	53	(9)	Ordinary
		Sell	2.336	(267)	Ordinary
		les	23	(9)	Ordinary
		les	8 750	(1000)	Ordinary
		9	3.281	(326)	Ordinary.
		100	10270	000	Guilary
		1120	400,1	(2 3)	Ordinary
		Sell	2,231	(255)	Ordinary
		Sell	735	(84)	Ordinary
		ll ex	840	(96)	Ordinary
		Sell	14,175	(1,620)	Ordinary
		Sell	19,031	(2.175)	Ordinary
		Seil	2.625	(300)	Ordinary
		la S	814	(63)	Ordinary
		19	175	(00)	Ordinary.
			0.70	(07)	oldina)
			000	(66)	Ordinary
		Seil	1,181	(135)	Ordinary
		Sett	1,890	(216)	Ordinary
		Sell	208	(28)	Ordinary
		Sell	350	(40)	Ordinary
Ť		Sell	21.254	(2.429)	Ordinary
		lles	590	(110)	Ordina
۳		197	506	015	Cicraly
April 4 Cos securires Australia Liu			1,024	(/)	Ordinary
1		yell .	19,031	(2,175)	Ordinary
		Sell	3,623	(414)	Ordinary
		Sell	64,846	(7,411)	Ordinary
3-Apr-14 UBS Securities Australia Ltd		Sell	3,071	(351)	Ordinary
i		lie de la la la la la la la la la la la la la	2 783	(318)	Ordinary
			ROT.	(5.5)	Ordinary
1		100	000	(30)	Cidrialy
1		100	4,218	(487)	Ordinary
1		Sell	14,149	(1,617)	Ordinary
CBS		iell	2,144	(245)	Ordinary
Apr-14 UBS Securities Australia Ltd		Sell	929	(75)	Ordinary
23-Apr-14 UBS Securities Australia Ltd		Sell	4.314	(493)	Ordinary
Ť		Call	77	(2)	Cardingo
T				C S	Ciciliary
T		Sell	S.	Ξ	Ordinary
3-Apr-14 UBS Securities Australia Ltd		Seli	929	(75)	Ordinary
		Seli	1,365	(156)	Ordinary
Apr-14 UBS Securities Australia Ltd		lle	929	(22)	Ordinary
۳		lie	959	(75)	Ordinan
23-Apr-14 I IRS Securities Australia 14			2 0	3 (5)	Croining
+	1		6 60 75	(L)	Ordinary
3-Apr-14 UBS Securities Australia Ltd		Seil	25,830	(2,952)	Ordinary
┪		Sell	656	(75)	Ordinary
_		Sell	929	(75)	Ordinary
3-Apr-14 UBS Securities Australia Ltd			219	(22)	Ordinary
t		وا	2636	(006)	Ordinary
Ť			520,2	(300)	Cigically
7		Sell	49,875	(5,700)	Ordinary
- 1		Sell	208	(28)	Ordinary
3-Apr-14 UBS Securities Australia Ltd		Sell	1,715	(196)	Ordinary
ı		119	1 838	(010)	Ordinary
			SCO'-	(0,12)	Guiraly

At was C	Carlotter Acceptance				/Cedipary
3-Apr-14	FIRS Securities Australia Ltd	EX 50	000	(0/)	Ciciliary
23-Anr-14	1 IRS Societies Australia Ltd	Coll	1014	(77)	Ordinary
14	II IBC Societies Australia I to	Soll	510,1	(01)	Ordinary
1 V V V C	Liber Committee Australia Ltd	llac llac	500	(9/)	Ordinary
17	I IBC Societies Australia 14d	Soll	10,1	(10)	Ordinary
23. Apr. 14	I IRC Coourition Australia 144	100	200,1	(110)	Cidinary
2 / 2 / 2	I IDS Countries Australia Lite	100	200,1	(1)	Ordinary
3-Apr-14	I IDC Countries Australia 144	Soll	2,045	(340)	Ordinary
3-Anr-14	LIRS Securities Australia 14d	Cell	2005	(0/1)	Ordinary
Apr. 14	I IDC Commission American Lead	250	2000	(040)	Countary
3.Apr-14	I IRC Committee Australia 14d	llos de la companya d	161	(599)	Ordinary
3.Apr.14	I IDC Counties Australia Ltd	300	450	(01)	Ordinary
3-Anr-14	LIRS Securities Australia Ltd	Coll	1,341	(904)	Ordinary
3-Apr-14	1 IRC Securities Australia Ltd	Soll	3,333	(400)	Ordinary
3.Apr.14	I IBC Countition Australia 1+d	Soli	4,990	(1 (2)(1)	Ordinary
Apr-14	I IRS Securities Australia 14d	Coll	12 130	(1,02.9)	Ciditaly
23.Apr.14	LIBS Securities Australia Ltd	Soll	13,123	(1,500)	Ordinary
1 2	Libe Securities Australia Ltd	24	901.0	(195)	Orginary
3-Apr-14	LIBS Securities Australia Ltd	Soll	0,730	(145)	Orginary
23. Apr. 14	LIRC Coo mitton Australia 14d	195	553,1	(140)	Ordinary
A April 1	LIBS Counties Australia Ltd	3611	23150	(1/4)	Ordinary
Apr. 14	LIBC Counties Australia 14d	list.	3,139	(36)	Ordinary
1 7 7 7 7	Tipe Committee Australia Ltd	liac liac	071	(01)	Orginary
1	Obs Securities Australia Lid	- N	607,1	(145)	Ordinary
Apr-14	USS Securities Australia Ltd	Ne:	73,643	(2,/02)	Ordinary
1 - Ide C	Obs Securities Abstralia Ltd	i k	104	(133)	Ordinary
5-Apr-14	UBS Securities Australia Ltd	Sell	809'7	(867)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	76,580	(8,752)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	2,485	(284)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	3,159	(361)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	4,104	(469)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	2,931	(332)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	5,136	(287)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	756	(1001)	Ordinary
23-Apr-14	UBS Securities Australia Ltd	Sell	5.110	(584)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	1 444	(165)	Ordinary
-Apr-14	LIBS Securities Australia Ltd	95	36 286	(2717)	Ordinan
3. Apr. 14	PR Contribut Australia Ltd	= 70	0.02,000	(000/	Cidillary
3-Apr-14	1.18C Counting Australia 14d	100	1,016	(370)	Oldinary
1 2	The Countries Australia cut	1120	200,1	0 10	Orginary
3-Apr-14	UBS Securities Australia Ltd	Sell	310,1	(116)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	2,030	(232)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	420	(48)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	376	(43)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	420	(48)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	420	(48)	Ordinan
3-Anr-14	I IRS Securities Australia 11d	3	2625	(006)	Ording)
2 4 2 4 4	I IDS Countries Australia Ltd	1100	2,023	(005)	Ordinary
1 1 1	ODS Securities Australia Ltd	Hac	420	(48)	Ordinary
+ 1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	ODS SECURITIES AUSTRAINA LIU	liac	3,045	(348)	Ordinary
3-Apr-14	OBS Securities Australia Ltd	Sell	358	(42)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	9,940	(1,136)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	368	(42)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	368	(42)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	884	(101)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	4,174	(477)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	18.095	(2.068)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Şei	2.030	(282)	Ordinary
3-Anr-14	I IRS Societies Australia Ltd	lle)	358	(42)	Ording
3-Apr-14	PRC Coording Australia 144	Coll	368	(42)	Ordinary
2	Coo Seculities Australia Liu	100	000	(47)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	1,015	(116)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	1,479	(169)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	2,783	(318)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	398	(42)	Ordinary
3-Apr-14	UBS Securities Australia Ltd	Sell	11,296	(1.291)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Seli	1015	(0.1)	Ordingo
		-	700	0	VICUIDIC)
23-Anr-14	II IDC Societifies Airetralia I td	100	898	(41)	Ordinary

UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd	1,383 1,383 1,383 1,234 1,234 1,234 1,150 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,374 1,389 1,189 1,	(15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3) (15/3)	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,383 2,599 1,1534 1,150 6,493 6,493 6,493 1,374 1,374 1,374 1,374 1,374 1,378 1,378 1,378 1,378 1,378 1,378 1,389	(1597) (1	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,599 1,123 1,123 1,123 6,433 6,433 1,274 1,374 1,374 1,374 1,374 1,2,189 1,3,189	(297) (360) (600) (77) (77) (77) (150) (10	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,234 1,234 5,506 6,493 6,493 6,493 6,493 1,150 1,	(141) (560) (77) (672) (673) (673) (673) (673) (62) (100) (634) (733) (733) (733) (733) (733) (733)	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,150 6,493 6,493 6,796 7,906 1,374 1,374 1,374 1,374 1,374 1,379 1,159	(350) (105) (105) (105) (105) (105) (105) (105) (105) (105) (105) (105) (105) (105)	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	525 6,493 6749 6749 1,374 1,374 1,374 1,374 1,378 1,588 1,588 1,2,189 1,2,189 1,2,189 1,2,189 1,2,189 1,2,189 1,2,189 1,3,189	(60) (72) (72) (72) (73) (150) (150) (150) (153) (153) (153) (153) (153) (153) (153) (153) (153) (153) (153)	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	6,433 6,433 2,66 5,906 1,374 1,374 5,548 5,548 5,548 2,049 3,159 1,2189 6,589 6,589 6,589 6,589 6,589 10,395 11,953 12,868 13,878 14,878 16,878 17,878	(72) (673) (150) (150) (100) (100) (100) (100) (100) (100) (130) (100) (130) (100) (Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,006 2,006 3,150 1,374 1,374 1,374 1,374 2,049 1,189 1,	(77) (97) (98) (100) (100) (100) (100) (100) (100) (100) (100) (100) (100) (100) (100) (100)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,906 3,150 1,374 1,374 1,374 1,374 2,049 1,218	(675) (360) (157) (157) (157) (157) (158) (158) (158) (105) (105) (105)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,049 1,374 1,374 1,374 5,448 1,2,189 1,2,	(360) (157) (157) (157) (100) (134) (1534) (1534) (1534) (1534) (1534) (1535) (1055) (1055)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,150 1,374 5,48 8,75 8,78 3,199 1,2,189 1,2,189 1,2,189 1,2,189 1,2,189 1,3,199	(157) (157) (100) (100) (100) (1234) (1234) (1234) (1235) (1235) (105) (105) (105)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,374 1,374 873 873 1,548 12,189 12,189 12,189 12,189 12,189 12,189 12,189 12,189 12,189 12,189 12,189 12,189 12,189 12,189 12,189 13,189 13,189 14,19 19,	(157) (100) (100) (234) (133) (133) (105) (105) (105) (105)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	873 875 875 875 876 876 877 877 877 877 877 877 877 877	(634) (100) (100) (100) (100) (100) (100) (100) (100)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,848 2,049 3,159 1,12189 6,589 6,589 6,589 1,419 1,039 1,039 1,953 1,95	(100) (634) (234) (134) (139) (139) (105) (105) (105) (105)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,548 2,049 1,189 1,189 1,189 1,189 6,589 4,419 1,199 1,250 5,250 5,250 1,250 1,397 1,968 1,968 1,968 1,968 2,748 2,748 2,748 2,748	(634) (234) (1,393) (439) (438) (438) (438) (438)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,159 12,189 3,789 6,589 6,589 10,295 10,205 10,205 10,205 10,205 10,205 10,205 10,205 10,205 10,205 10,205 10,205 10,205 10,205	(234) (1,393) (433) (105) (105) (438)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,789 3,789 6,589 6,589 4,419 3,833 3,833 5,250 5,250 5,250 6,250 6,403 1,963 1,	(1,393) (1,393) (433) (505) (605) (605)	Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	12,189 3,789 6,589 4,19 3,833 5,250 5,250 5,250 10,395 11,953 11,953 11,953 11,953 11,953 11,953 11,953	(1,393) (433) (753) (505) (105) (438)	Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,788 6,589 4,419 919 919 5,250 10,280 1,580 4,568 4,568 1,569 1,563 1,1953 1,1	(433) (753) (505) (105) (438)	Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	6,589 4,19 3,833 3,833 3,833 5,250 5,250 10,395 4,568 4,568 4,568 4,568 4,568 2,748 2,748 2,748	(753) (505) (105) (438)	Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	4,419 1,839 1,839 1,839 1,250 1,395 1,953 1,953 1,953 1,953 1,953 1,953 1,953 1,953 1,953 1,953 1,953 1,953 1,953	(505) (105) (438)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	9 19 9 9 19 9 19 9 19 9 19 9 19 9 19 9	(438)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,833 5,250 5,250 10,395 4,568 3,876 11,953 7,79 49,738 2,748 2,748	(438)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,250 10,395 10,395 4,568 3,876 11,953 49,79 49,438 2,748 2,748	(009)	Committee
UBS Securities Australia Ltd UBS Securities Australia Ltd	10,395 10,395 4,568 3,876 11,953 779 49,438 2,748	(000)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	10,395 4,568 3,876 11,953 779 49,488 2,748	(009)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	4,568 3,876 11,953 779 49,438 2,748 2,748	(1,188)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,876 11,953 779 49,438 2,748	(522)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	11,953 779 49,438 2,748 534	(443)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	779 49,438 2,748 534	(1,366)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,748 534	(88)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,748	(2,650)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	534	(314)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	-	(61)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,09,1	(183)	Ordinary
UBS Securities Autoralia Ltd UBS Securities Autoralia Ltd	17,351	(1,983)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	999	(92)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1,120	(128)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1,146	(131)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	929	(22)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	200	(08)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	4,760	(544)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	674	(77)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1,365	(156)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	999	(92)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	8,260	(944)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	929	(75)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	9.888	(1.130)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	999	(19/	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	2,214	(253)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	22,645	(7.588)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	949	(75)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2 380	(070)	Ordinary
UBS Securities Australia Ltd	1,654	(180)	Ordinary
Obs securities Australia Liu	109.0	(1 108)	Countries
23_Apr.14 IBS Converties Australia 1td	1 034	(1,100)	Ordinary
LIDE Committee Australia 144	100	(32)	Ordinal y
5-Api-14 Libs securities Australia Lid	600	(0/)	Ordinary
Tibe counties Australia Lid	0,423	(724)	Ciginaly
UDS Securities Australia Ltd	500	(q/)	Ordinary
UBS Securities Australia Ltd	56,648	(6,474)	Ordinary
-1	9,380	(1,0/2)	Ordinary
UBS Securities Australia Ltd	718	(82)	Ordinary
3-Apr-14 UBS Securities Australia Ltd Sell	1,015	(116)	Ordinary
	999	(22)	Ordinary
	683	(78)	Ordinary
	500	000	

ODS Securities Australia Ltd	Seli	800	/75/	Oromary
UBS Securities Australia Ltd	Sell	1,015	(116)	Ordinary
UBS Securities Australia Ltd	Jack Spell	4 060	(444)	Ordinary
JBS Securities Australia Ltd	3	3.640	(416)	Ordinary
UBS Securities Australia Ltd	3	368	(CV)	Ordinary
BS Sequirities Australia Ltd	9	109	(97)	Croinary
UBS Securities Australia Ltd	les	8.750	(1 000)	Ordinary
JBS Securities Australia Ltd	Sell	368	(42)	Ordinary
JBS Securities Australia Ltd	Sell	1.006	(115)	Ordinary
UBS Securities Australia Ltd	Sell	368	(42)	Ordinary
UBS Securities Australia Ltd	Sell	368	(42)	Ordinary
UBS Securities Australia Ltd	Sell	875	(100)	Ordinary
UBS Securities Australia Ltd	Sell	368	(42)	Ordinary
JBS Securities Australia Ltd	Sell	1,015	(116)	Ordinary
UBS Securities Australia Ltd	Sell	3,045	(348)	Ordinary
JBS Securities Australia Ltd	Sell	368	(42)	Ordinary
JBS Securities Australia Ltd	197	σ	(1)	Ordinary
UBS Securities Australia Ltd	97	368	(04)	Ordinary
JBS Securities Australia 11d	100	896	(42)	Ordinary
IRS Securities Australia 14d	100	11 305	(1 202)	Ordinary
I IBC Coo rition Australia 14d	lies lies	2001	(2671)	Oldinary
Obsider Answerig to	Nell Nell	010,1	(01)	Ordinary
Obsisted High Australia Ltd	Sell	510,1	(911)	Ordinary
UBS Securities Australia Ltd	Sell	368	(47)	Ordinary
UBS Securities Australia Ltd	Sell	3,500	(400)	Ordinary
UBS Securities Australia Ltd	Sell	11,961	(1,367)	Ordinary
JBS Securities Australia Ltd	Sell	10,780	(1,232)	Ordinary
UBS Securities Australia Ltd	Sell	12,023	(1,374)	Ordinary
JBS Securities Australia Ltd	Seli	4,865	(226)	Ordinary
JBS Securities Australia Ltd	Seli	4,375	(200)	Ordinary
JBS Securities Australia Ltd	Sell	5,381	(615)	Ordinary
JBS Securities Australia Ltd	lles	208	(28)	Ordinary
JBS Securities Australia Ltd	Sell	3,623	(414)	Ordinary
UBS Securities Australia Ltd	Seli	543	(62)	Ordinary
JBS Securities Australia Ltd	Sell	648	(74)	Ordinary
UBS Securities Australia Ltd	Sell	3,168	(362)	Ordinary
JBS Securities Australia Ltd	Sell	42,796	(4,891)	Ordinary
UBS Securities Australia Ltd	Sell	2,546	(291)	Ordinary
UBS Securities Australia Ltd	Sell	208	(28)	Ordinary
JBS Securities Australia Ltd	Sell	5,933	(678)	Ordinary
UBS Securities Australia Ltd	Sell	368	(42)	Ordinary
UBS Securities Australia Ltd	Sell	2.030	(232)	Ordinary
UBS Securities Australia Ltd	lleS.	368	(42)	Ordinary
URS Sequirities Australia 11d	3	3.684	(177)	Ordinary
JBS Securities Australia 11d	3	674	(77)	Ordinary
IRC Securities Australia 14d	Total	300 /	((//)	Ordinary
USC Consister Assessing to	ll Sc	303 0	(500)	Ordinary
IRS Securities Australia 11d	loo!	177 5	(300)	Ordinary
I IRS Securities Australia 1td	190	1 7/5°	(122)	Ordinary
IBS Securities Australia Hd	3	166	(101)	Ordinary
JBS Securities Australia 11d	9	368	(64)	Ordinary
UBS Securities Australia I td	lia)	13 676	(1 563)	Ordinary
UBS Securities Australia Ltd	Sell	4,060	(464)	Ordinary
UBS Securities Australia I td	la S	5.066	(579)	Ordinary
UBS Securities Australia Ltd	Şe	1.575	(180)	Ordinary
	Setl	1,313	(120)	Ordinary
UBS Securities Australia Ltd	Sell	4,051	(463)	Ordinary
UBS Securities Australia Ltd	Sell	398	(42)	Ordinary
UBS Securities Australia Ltd	Sell	1,015	(116)	Ordinary
UBS Securities Australia Ltd	Sell	3,045	(348)	Ordinary
UBS Securities Australia Ltd	Seil	1,208	(138)	Ordinary
UBS Securities Australia Ltd	Sell	368	(42)	Ordinary
JBS Securities Australia Ltd	Sell	4,060	(464)	Ordinary
JBS Securities Australia Ltd	lles.	4,051	(463)	Ordinary
UBS Securities Australia Ltd	Sell	368	(42)	Ordinary
JBS Securities Australia Ltd	1143			
	1110	2.030	(232)	Ordinary

Γ	Τ							[Τ	T	Τ	Τ	Τ	Ī	Ī					Τ	Τ	T	Ι				1		T			<u> </u>	П	_		1		T	1	I	Τ	Τ							_
Ordinan	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary																			
(1)	(42)	(81)	(42)	(42)	(42)	(42)	(42)	(42)	(42)	(42)	(464)	(238)	(1,500)	(009'9)	(234)	(448)	(5,001)	(36)	(427)	(A)	(95)	(1,159)	(140)	(325)	(1,874)	(159)	(362)	(21)	(748)	(115)	(8/6/9)	(103)	(311)	(1,052)	(683)	(47)	(888)	(363)	(5,468)	(362)	(414)	(122)	(1,807)	(5,134)	(202)	(107)	(42)	(114)	(42)	(463)	(116)	(116)	(232)	(158)	(42)	(197)	(116)	(116)	(42)	(247)
σ	368	709	368	368	368	368	896	368	398	368	4,060	2,083	13,125	052,750	2,048	3,920	43,759	315	3,736	1,741	371 NS	10 141	1 225	2,844	16,398	5,521	3,168	184	6,545	1,006	6 151	100	2.721	9,205	4,716	411	7,77	3,176	47,845	97,108	3.625	1,068	15,820	44,923	1,769	986	200	866	368	4,051	1.015	1,015	2,030	1,383	368	1,724	1,015	1,015	368	2.16
Self	Sell	Setl	Sell	Seil	Seli	Sell	Sell	Sell	Sell	Sell	Ne de la constant de	Sell	lle y	Jies	Sell	Sell	Sell	Sell	Seli	Sell	Sell	Je Je	Jan Jan Jan Jan Jan Jan Jan Jan Jan Jan	- N	Sell	Sell	Sell	Sell	Sell	les c	Zeil	Sell	Sell	Sell	 8	Zell	le le		į į	3	Sell	٥																		
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	USS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	USS Securities Australia 11d	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IRS Securities Australia Ltd
23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	.23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	23-Apr-14	П	T	T	23-Apr-14				Ī	Ī	Ť	23-Apr-14																

1 V	COST SECURISES AUSTRAINS LICE	200	3.447.74		Crainary
	LIBS Sequitities Australia Ltd	Blac	517 216	300 03	Coling
4-Apr-14	UBS Securities LLC	Stock returned	ANN	(3.725)	Ordinary
24-Apr-14	I IRS Securities 110	Ctock rottimod	4714	(627,0)	Carling,
24-Apr-14	I IRC Contrition Australia 14d	Contracting Desiration	X21	(401)	Ordinary
24-05-14	LIPO Cocuration Australia Lea	Cotte and Delivering	(A)	307	Coprioris
24-Apr-14	LIBC Constitute Australia 144	Cater into Deliverine	E	2 2	Contons
1	The Manual Manual and Line	בוויבו וווס ביפוואפוואפ	A/N	Onc.	Options
11.00-1-1	Coo wealth Management Australia Ltd	ng.	4,155	744	Ordinary
-10-1	ODS Securities Australia Ltd	Nell Nell	1,024	(116)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	18	(2)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Jag.	8,830	(1,000)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	17,510	(1,983)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	31,347	(3,550)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Seil	1,766	(200)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	6,623	(220)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	26	(3)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	6/8/9	(6/2)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	2,960	(675)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	35	(4)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	15,479	(1,753)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	les	1,757	(199)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	ay.	108 706	(112 311)	Ordinary
4-Apr-14	UBS Securities Australia I td	(19)	AE1	(13)	Ordinary.
4-Apr-14	UBS Securities Australia Ltd	las	1 148	000	Croins
1-Apr-14	UBS Securities Australia I td	و	6,879	(05.1)	Crainary
4-Apr-14	UBS Securities Australia I td	Z 2	56.441	(6 302)	Ordinary Training
4-Apr-14	UBS Securities Australia Ltd	(a)	1111	(127)	Ordinary
4-Apr-14	UBS Securities Australia I td	9	3 161	(328)	Caling
4-Apr-14	I IBS Societies Australia 14d	100	101,0	(202)	1000
4-Anr.14	I IRC Securities Australia 14d	lac o	0300	(670)	Oldinary
4 4 7	I DC Complete Australia Ltd	11-0	606'6	(1,123)	Ordinary
#1-IdV-#	ODS Securities Australia Ltd	Seil	5	Ê	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Jes	10,381	(1,173)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Seil	1,027	(116)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Seil	1,027	(116)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	2,053	(232)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	75,497	(3,107)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	35	(4)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	1,036	(117)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	1,425	(161)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	6 204	(701)	Ordinary
4-Apr-14	UBS Securities Australia I to	(a)	1 378	(150)	Ordina
A Apr. 14	Hoo Cocurition Australia 144	107	255,	(000)	Oldinary
1-Idw-14	ODS Securities Australia Lita	Sell	9/5/1	(1,986)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	246,057	(27,803)	Ordinary
1-Apr-14	UBS Securities Australia Ltd	Sell	7,053	(797)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	2,655	(300)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	3.168	(358)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	les	248	(28)	Ordinary
1-Anr-14	I IBS Securities Australia I to	Coll	1 637	(301)	or or or or
A Apr 14	1 DC Cocurition Australia Ltd	1150	1,503,1	(201)	Colling
1000	ODS Securities Australia Ltd	lac lac	3,131	(956)	Ordinary
4-Apr-14	UBS Securities Australia Lto	Sell	1,221	(138)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	9,992	(1,129)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	6,213	(202)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	1,142	(129)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	2 443	(976)	Ordinary
4-Anr-14	I IBS Securities Australia 1 td	Coli	1	(1)	Ordings
4-Apr-14	I IRS Securities Australia 14d	Coll	1 303	/1 /0 //	ording.
	100 Complete Author Con	100	502,1	(010)	Cicilary
4-MDI-14	Los securites Australia Ltd	II I	15,/44	(6//1)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Nell Nell	3,151	(326)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	230	(56)	Ordinary
I-Apr-14	UBS Securities Australia Ltd		38,984	(4,405)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	les.	2,425	(274)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	18	(2)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sel	7.080	(800)	Ordinary
ı	UBS Securities Australia Ltd	Set	6.638	(750)	Ordinary
					,
	It IRC Con without Australia ! td	- J	717	(181)	Ordinany

Γ	П					Γ			Γ		Γ	Τ		Τ	I			I						_	Т		1	1	Τ		T	Τ		Γ	Γ				T	T	T	T	1					Т	T	Τ	Τ	Т		-		П	7	1	Т	Т	Ī
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary
(42)	(42)	(2,980)	(3,460)	(4,429)	(1,207)	(969'1)	(234)	(95)	(62)	(139)	(3.471)	(169)	(361)	(4 075)	(105)	(19)	(1 600)	(682)	(19)	(637)	(61)	(325)	(715)	(2,000)	(200)	(000/1)	(9/0/1)	(2)	(191)	(972)	(995)	(144)	(460)	(2)	(1,716)	(200)	(23)	(168)	(11,548)	(87)	(5,000)	(174)	(105)	(193)	(622)	(2)	(361)	(186)	4,695	(149,629)	(193)	(133)	(11,507)	(13,332)	(2)	(120)	36,360	(5,496)	13,064	(4,516)	10,002)
368	398	26,075	30,275	38,754	10,561	14,840	2,048	490	831	1,216	30.371	1 479	3.159	35 656	919	156	14 000	2,968	166	5,574	166	2,845	6,260	43,750	1,751	8,750	9,415	18	1/9/1	278	8 711	1,260	4,027	81	15,015	1,750	464	1,470	101,045	1 705	7507,1	1.085	916	1,689	6,816	18	3,159	1,628	41,478	1,319,636	VAN	A N	AN A	ΝΑ	NA	NA	320,899	48,485	115,294	39,921	27.200
Sell	Seli	Sell	Je.	Sel	la S	Je Je	, le	J. J.	Sell	Sell	Seil	Sell	Sell	Sell	Sell	Sell	Sell		les l		Jac.	3	llay.	lles.	Sell	Sell	Sel	Self	Sell	Sell	la Sell	į į	S S	Sell	Sell	Sell	Sell	Sell	Sell	Buy	Sell	Stock returned	Buy	lles.	Buy	Jes Jes	100														
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia I td	UBS Securities Australia IId	UBS Securities Australia 11d	UBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia 1.td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG Australia Branch	UBS AG AUStralia Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS AG London Branch	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Obsiderations Australia Liu
23-Apr-14					23-Apr-14		23-Apr-14	П	Г	Г	T	T	T	1	23-Apr-14	1			Ť	Г	23-Apr-14			- [- 1	- 1	23-Apr-14	1	Ť	23 Apr 14	T	23-Apr-14	Ť	Ī	Ħ		23-Apr-14	-	7	23-Apr-14	T	T	23-Apr-14			П	23-Apr-14	T	+	T	T	24-Apr-14	Ť	Ť	Ħ	Ť	T	7	T	T	**-Apr-42

	1. 10 C C	- 0			Ordinan,
24-Apr-14	UBS Securities Australia Ltd	- S	510,71	(1,92,1)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	iac s	58,084	(8/5,0)	Ordinary
# - Id	ODS Securities Australia Ltd	Sen Sen	000	(89)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	31,947	(3,618)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sel.	25,828	(2,925)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	1,086	(123)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	1.581	(179)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	53	(9)	Ordinary
24-Apr-14	I IRS Securities Australia I to	Joj	25 020	(3000)	or district
24-Apr-14	I IRS Securities Australia 11d	Coll	100,000	(5,700)	in the contract of
	Lab Consider Assessing	370	261,10	(00,1,0)	9 9
4-MDI-14	Doo Securities Australia Ltd	Sell	4,415	(200)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	40,936	(4,636)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	12,044	(1,364)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	371	(42)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	1,272	(144)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	3.567	(404)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	52 373	(4 917)	Ordinary
-Anr-14	UBS Securities Australia I td	lo ₂	5,510	(45.7)	, acripio
Apr-14	I IBS Securities Australia I to	100	606.06	(07,0)	Colling
A Apr 14	I IDC Counties Augustia 14d	liec J	CECOC	(2,442)	G Giraly
1	Discourages Australia Literatura	nac nac	006	(901)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Seil	42,110	(4,769)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	5,298	(600)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	177	(20)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	87,346	(6,892)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	£20'2	(801)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	028'99	(7.573)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	leS	14 305	(1,620)	Ordinary
4-Anr-14	UBS Securities Australia I to	le y	9636	(111)	Ordingo
A Apr 14	1.00 Cooperation Australia (+)	220	2440	(555)	Ordinary Ordinary
1	ODD Securities Australia Lid	136	2,440	(1/7)	Ordinary
4-4DI-14	ODS SECULITIES AUSTRAINA LIG	Sell	859,1	(1,318)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	46,243	(5,237)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	1/2	(8)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	22.543	(2.553)	Ordinary
4.Ans.14	II IBS Societies Australia I to	Coll	09064	(1001)	Carding
Apr. 14	I IRC Securities Australia I to	=00	200,01	(103)	Ordinary California
:	יייים ביייים וופס עמיוויים דות	100	0101	(501)	Ordinary
54-Apr-14	Obs securities Australia Ltd	Sell	94	(77)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	715	(81)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Se.	56	(3)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	195	185	(12)	Ordinary
1-Apr-14	I IBC Committee Ametralia 144	100	0000	(120)	1
+1-Idw-	ODS Securities Australia Lid	liac	3,108	(355)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	627	(71)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	6	Ξ	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	1.236	(140)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	ay.	5 731	(649)	Ordinary
Apr 14	I IDC Contration Automotive Lead	100	1000	(60.5)	(and)
4-ADI-14	Obs Securities Australia Ltd	Sell	1,448	(154)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Seli	8,371	(948)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	459	(25)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	371	(42)	Ordinary
24-Anr-14	UBS Securities Australia Utd	Sell	1 713	(104)	Ordinan
1. Apr. 14	1 ISC Countition Australia 14d	==0	0.00	(100 +)	
1	COO Securities Australia Ltd	3611	14,930	(+60'1)	Collinally
4-Apr-14	UBS Securities Australia Ltd	Sell	2,437	(276)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	3,720	(421)	Ordinary
-Apr-14	IUBS Securities Australia Ltd	Sell	3,965	(449)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	29E	(41)	Ordinary
4 Apr 14	At Citation A solition Co. DOI -	100	0000	(0000)	(5)
1	DDD Securities Australia Ltd	IBG I	670'61	(2,2,2)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	6	(E)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	4,647	(256)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	2,649	(300)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	lleS	174	(31)	Ordinary
4-Apr-14	1 BC Countition Assets 14d		1004	(44.5)	Grindy
1	Opp Securities Australia Ltd	Jell Jell	1,034	(117)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	168	(19)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	13,703	(1,551)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	221	(52)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	lleS	132	(15)	Ordinary
17 1 1 1 1	Il Ind Constitution Assets in 140	11.0	090 1	(222)	
24-Apr-14	UBS Securities Australia Ltd	Self	2,960	(675)	Ordinary

4-Apr-14 4-Apr-14 4-Apr-14 4-Apr-14 4-Apr-14	UBS Securities Australia Ltd	Sell	3,151	(356)	Ordinary
Apr-14 Apr-14 Apr-14 Apr-14			305 1	(464)	Ordinary
Apr-14 Apr-14 Apr-14 Apr-14	UBS Securities Australia Ltd	Sell	4,106		
Apr-14 Apr-14 Apr-14	UBS Securities Australia Ltd	Sel	3,152	(357)	Ordinary
Apr-14	UBS Securities Australia Ltd	Set	6,623	(750)	Ordinary
Anr-14	UBS Securities Australia Ltd	Seli	203	(23)	Ordinary
	UBS Securities Australia Ltd	Sell	124	(14)	Ordinary
	UBS Securities Australia Ltd	Sell	1,943	(220)	Ordinary
- [UBS Securities Australia Ltd	Zell	/79	(71)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	717	(24)	Ordinary
- [UBS Securities Australia Ltd	- R	3,161	(358)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	- Rel	3,161	(358)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	ner.	6 5		Ordinary
ł	Upo Securities Australia Liu	nac Sen	4//	(40)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sel	1,740	(197)	Ordinary
- 1	UBS Securities Australia Ltd	Seil	24/	(28)	Ordinary
1	UBS Securities Australia Ltd	Sell	8,247	(934)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Zei	2,358	(267)	Ordinary
-1	UBS Securities Australia Ltd	Sell	3,152	(755)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	3,152	(35/)	Ordinary
-	UBS Securities Australia Ltd	Sell	7/1/7	(240)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	88,300	(000,01)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	689'91	(068'1)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	3,152	(357)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	1,183	(134)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	3,161	(358)	Ordinary
- 1	UBS Securities Australia Ltd	Self	5,298	(009)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	11,638	(1,318)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	230	(09)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	902	(782)	Ordinary
	UBS Securities Australia Ltd	Sell	2,773	(314)	Ordinary
	UBS Securities Australia Ltd	Sell	23,846	(5,699)	Ordinary
1	UBS Securities Australia Ltd	Sell	250'2	(662)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	3,152	(357)	Ordinary
1	UBS Securities Australia Ltd	Sell	3,161	(358)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	6.525	(739)	Ordinary
4-Anr-14	I IRS Securities Australia I to	(a)	VC1 8	(020)	Ordinary
1,1	U.D.C. Completes Australia 144		1200	(076)	Ciditaly
- Idu-	ODS Seculiales Australia Ltd	120	607,1	(140)	Ordinary
4DI-14	UBS Securities Australia Ltd	Ser.	8,044	(16)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	lleci	795	(41)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	1,104	(125)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	3,161	(358)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	8,989	(1,018)	Ordinary
^pr-14	UBS Securities Australia Ltd	Sell	1,192	(135)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	8,521	(962)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	44,150	(2,000)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	ll as	18,102	(2.050)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Se l	6.826	(773)	Ordinary
Apr-14	URS Securities Australia 14d	ق ک	10.146	(0/1)	Ordinary
7 7 7 7 7	LIDE Commission Automatical and	#25	C1C1	(621)	Ordinary
1	ODS Securities Australia Liu	250	Z#C'	(701)	Granary
24-Apr-14	UBS Securities Australia Ltd	Sei	867'9	(909)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	7,143	(808)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	168	(19)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	3,161	(328)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	17,660	(2,000)	Ordinary
24-Apr-14	UBS Securities Australia Ltd	Sell	204,697	(23,182)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	6	(5)	Ordinary
24-Anr-14	UBS Securities Australia I td	la.	3.152	(357)	Ordinary
4-Anr-14	I IRS Securities Australia 14d	8 9	0	(1)	Ordinary
4-Apr-14	I IRC Societies Australia Ltd	lik lik	078 3	(022)	Croins
1	ODS Securities Australia Liu	343	0,000	(6//)	Clairiary
1	UBS Securities Australia Ltd	Sell	511	(61)	Ordinary
1	UBS Securities Australia Ltd	Sel	2,720	(308)	Ordinary
	UBS Securities Australia Ltd	Sell	696'6	(1,129)	Ordinary
4-Apr-14	UBS Securities Australia Ltd	Sell	6/8/9	(272)	Ordinary
Apr-14	UBS Securities Australia Ltd	Sell	686	(112)	Ordinary
24-Apr-14	LIRS Securities Australia Ltd	Sell	686	(112)	Ordinary
Т	U.D. Commission Australia Ltd	100	245	(211)	Ordinal y

24-Apr-14 UBS Securities Australia Ltd 24-Apr-14 UBS Secu			22,949 11,258 8,830 2,745 9 9 2,066 6,834 6,834 11,550 11,550 19,029 2,739 1,734 1,734 1,1	(1,259) (1,279) (1,279) (1,279) (1,279) (1,279) (2,111) (2,111) (2,111) (2,111) (2,111) (2,111) (2,111) (2,111) (3,11) (4,11) (4,11) (4,11) (4,11) (6,11) (1	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			2.2,449 11,256 8,830 2,746 6,834 11,550 19,029 2,799 2,799 18,640	(1.275) (1.275) (1.275) (1.275) (1.1000,1) (1.275) (1.100,1) (1.275) (1.11,1) (2.24,1) (2.24,1) (3.175	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1			8,830 459 459 6,834 9 11,550 19,029 2,739 2,739 1,113	(1,000) (1,000	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			2,056 2,066 2,066 2,066 2,066 2,066 11,550 11,550 11,029 2,1342 2	(52) (53) (62) (1) (62) (1) (1) (1) (1) (1) (2) (1) (1) (2) (3) (4) (4) (4) (4) (4) (5) (62) (1) (7) (62) (1) (7) (7) (8) (8) (8) (8) (8) (8) (8) (8) (8) (8	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			2.74 459 6.834 6.834 11,550 19,029 2,739 2,739 1,13 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,13 1,143 1,1	(15) (17) (17) (17) (17) (17) (17) (17) (17	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			459 684 684 684 11,550 11,550 19,029 2,799 2,799 2,799 2,810 8,640 18,640 18,640 18,640 18,640 18,640 18,640 18,640 18,640 18,640 19,810 1,113 1,113 1,113 1,113 1,517 1,113 1,520 1,52	(1) (1) (234) (1) (234) (234) (234) (2412) (Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			2,006 6,834 6,834 11,550 19,029 2,734 2,739 4,344 1,640 1,840 1,13 1,13 1,13 1,13 1,13 1,13 1,13 1,1	(1) (1) (234) (1) (1) (234) (1) (234	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			2,066 6,834 11,550 19,029 2,739 4,344 1,844 1,113 1,11	(12) (12) (12) (13) (14) (15) (17) (17) (17) (17) (17) (17) (17) (17	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			6834 11,550 11,550 19,029 2,799 2,799 1,840 1,840 1,841 1,113	(77) (77) (1,308) (2,155) (3,17) (3,17) (49) (49) (27) (41) (1,11) (27) (27) (48) (48) (48) (48) (48) (57) (1,11) (1,11) (27) (1,11) (27) (1,11) (27) (1,11) (1,28)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			11,550 19,029 2,399 2,799 1,344 18,640 18,640 2,881 1,113 1,	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
			11,550 19,029 21,342 1,344 18,640 18 9,810 238 35 7,717 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,113 1,279 15,779 15,779 16,779	(1308) (377) (377) (377) (377) (377) (378) (48) (48) (48) (58) (58) (68) (75) (75) (75) (75) (75) (75) (75) (75	Ordinary Ordinary Ordinary Ordinary
]			19,029 2,794 2,794 18,40 18,640 18,640 2,810 35 35 35 37 113 1,113	(2.15.) (3.17) (3.17) (43.2) (43.2) (43.2) (4.11) (2.7) (1.11) (2.7) (1.11) (2.7) (1.12) (1.13) (1.14) (2.7) (1.15) (1.16) (1.17) (1.16) (1.17	Ordinary Ordinary Ordinary
]			21,342 2,799 2,799 1,344 1,844 9,810 2,38 3,5 7,777 7,777 1,113 1,113 1,113 1,5 1,5 1,13 1,13 1,1	(32) (432) (432) (432) (2) (1111) (27) (136) (13	Ordinary Ordinary Ordinary
			2.799 4.344 18.640 18.640 238 35 7.717 1,113 44 15.779 86,755 9,969 9,969 9,969 1,864 1,864 1,864 3,188	(11) (11) (11) (11) (11) (11) (11) (11)	Ordinary
1			18 640 18 640 18 640 288 288 288 7 7 7 7 1,113 159 662 662 662 86,755 86,755 86,755 1,864	(492) (111.1) (2) (11.1) (2) (11.1) (2) (1) (1) (1) (2) (1) (2) (1) (1) (2) (2) (2) (3) (3) (4) (8) (8) (8) (8) (1) (1) (1) (1) (1) (1) (2) (3) (4) (4) (5) (6) (6) (6) (7) (7) (7) (7) (8) (8) (7) (8) (8) (8) (9) (9) (9) (9) (9) (9) (9) (9) (9) (9	Ordinary
1			18,640 18 18 28810 28810 2787 7,777 1,113 44 159 66,755 86,755 86,755 9,969 9,969 9,969 1,854 1,1884	(111, (2) (2) (2) (3) (4) (4) (4) (4) (4) (4) (4) (4) (4) (4	Crdinary
			15,000 18 9,810 238 238 238 24 1,113 1,113 1,113 662 662 662 662 662 662 662 662 662 66	(11,17) (11,17) (21,17	280
			9,810 238 238 55 7,717 1,113 44 1159 662 662 15,779 86,755 86,755 9,969 9,969 9,969 1,884	(111.1) (111.1) (12.1) (13.2) (13.2) (13.2) (13.2) (13.2) (13.2) (13.2) (13.2) (13.2) (13.2) (13.2) (13.2) (13.2)	200
			9.810 2.83 7.77 7.77 1,113 44 159 662 15,79 86,755 86,755 9,969 9,969 9,969 1,854 1,854 1,854	(17) (126) (126) (126) (126) (136) (136) (136) (136) (136) (137) (137) (138) (Ordinary
			288 288 35 1,113 1,113 662 662 662 15,779 86,755 9,969 9,969 1,969 1,864 1,864	(874) (126) (126) (127) (137) (187) (1,75) (1,75) (1,75) (1,129) (1,129) (1,129) (1,129)	Ordinary
			7,717 1,113 1,113 159 162 662 15,779 86,779 9,969 9,969 9,969 1,884	(4874) (1806) (1807) (1807) (1817) (1787) (1787) (1787) (1787) (1787) (1787) (1787) (1787)	Ordinary
			7,717 1,113 44 159 662 15,79 86,755 9,969 9,969 1,854 1,1854	(874) (126) (18) (18) (175) (1787) (1787) (1787) (1787) (1787) (1729) (1729) (1729)	Ordinary
			1,113 44 159 662 662 1,5779 86,755 9,969 9 1,884 1,188	(126) (5) (18) (75) (1,787) (9,825) (1,129) (1,129) (1,129)	Ordinary
			159 662 662 15,779 86,755 9,969 9 1,854 1,188	(5) (18) (75) (1,787) (9,825) (1,129) (1) (1)	Ordinary
			159 662 15,779 86,755 9,969 9 9 1,854 1,884	(18) (75) (757) (9,825) (1,129) (1) (1) (1) (1)	Ordinary
1			662 15,779 86,755 9,969 9 1,854 3,188	(757) (757) (9,825) (1,129) (1) (1)	Ordinary
			15,773 15,773 86,755 9,869 9 1,854 3,188	(5,787) (9,825) (1,129) (1) (210)	in diameter
			86,755 9,969 9,1854 1,854	(9,825) (9,825) (1,129) (1) (1)	Cigniary
			9,969 9,969 9 1,854 3,188	(1,129) (1,129) (1) (210)	Oldinary -
			9,969 9 1,854 3,188	(1,129)	Ordinary
			9 1,854 3,188	(210)	Ordinary
			1,854	(210)	Ordinary
			3,188	1190)	Ordinary
			200.12	114	Ordinary
		=	F 474	(100)	6
			3,1/4	(000)	Calnary
			105,1	(0/1)	Ordinary
		Sell	300	(34)	Ordinary
		16	3,956	(448)	Ordinary
			7,179	(813)	Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		-	8 901	(1 008)	Ordinary
1 []] [] [] [] [] [] [] [] []		Carl	1 236	(000)	(minute)
			1,230	(040)	Comary
			108,101	(11,529)	Ordinary
			1,325	(120)	Ordinary
		Sell	5,413	(613)	Ordinary
		Sell	7.029	(962)	Ordinary
		100	CN0 2	(571)	Ordinan
		100	2100	(1, /C)	O Collinary
			ns l	(71)	Ordinary
			44	(2)	Ordinary
		Sell	18	(2)	Ordinary
1 1 1 1 1 1 1			62	(2)	Ordinary
1 1 1 1 1		70	44	(5)	Ordinary
1111			63	G (9	Carion
1 1 1 1			2000	(0)	or and or
111			7,281	(907)	Orginary
11	-td		1,025	(116)	Ordinary
1		-	1,025	(116)	Ordinary
ı		Selt	1.025	(116)	Ordinary
		المحال	1 025	(116)	Grain
Ť			1,035	(310)	in in
Ť			1,023	(00,1)	d diam
T			9,724	(1,100)	Ordinary
4-Apr-14 UBS Securities Australia Ltd		Sell	2,051	(232)	Ordinary
			1,025	(116)	Ordinary
4-Apr-14 UBS Securities Australia Ltd	td Sell		35	(4)	Ordinary
Г			4 906	(555)	Ordinary
T			47.300	(2007)	da li
T			42,500	(4,700)	Ordinary
		Sell	16,858	(1,907)	Ordinary
4-Apr-14 UBS Securities Australia Ltd		-	1,450	(164)	Ordinary
Т	leS to	-	9.892	(1.119)	Ordinary
Г		11'	14 338	(1,622)	Ordinan,
24-Api - 14 Obs securities Australia Ltd	Tec.		14,338	(1,522)	Ordinary

3,929 124 12,795 1,080 2,870 2,870 2,870 1,439 1,124 1,134 1	(445) (149) (120) (325)	Ordinary Ordinary Ordinary
12.795 1.080 2.870 2.870 2.870 3.948 1.143 1.1439 1.1,347 1.1,	(1,449) (1,449) (120) (325)	Ordinary
1,080 1,080 2,870 2,870 3,974 1,374 1,374 1,374 1,343 1,343 1,174	(1,449)	Ordinary
2,870 2,870 5,448 3,974 1,439 1,134	(120)	
2,870 2,480 1,439 1,139 1,134	(325)	Ordinary
3,448 3,974 1,439 1,439 1,1,947 11,947 1,174	10.00	Ordinary
9,907 1,439 11,947 11,1947 1,174 1,1	(/19)	Ordinary
9,907 1,439 1,1947 11,947 11,194 1,174 1,174 1,174 1,174 1,174 1,174 1,174 1,174 1,174 1,174	(450)	Ordinary
1,439 1,449 1,144 1,146 1,174 1,174 1,339 1,339 1,8084	(14)	Ordinary
1,449 1,947 9,466 1,1717 1,171 1,174 1,178	(1,122)	Ordinary
11,947 9,466 1,174 3,329 18,084	(163)	Ordinary
9,466 1,174 3,329 18,084	(1,353)	Ordinary
1,174 3,329 18,084	(1,072)	Ordinary
3,329	(133)	Ordinary
18,084	(777)	Ordinary
100,01	(0/0/0/	Ordinan
	(4) (4)	1000
6/6/61	(4,134)	Ordinary
4,424	(104)	Ordinary
14,728	(1,668)	Ordinary
9,360	(1,060)	Ordinary
19,779	(2,240)	Ordinary
2 305	(1961)	Ordinary
000,31	(103)	Cidinary
05,20	(660,1)	Oromany
4,777	(541)	Ordinary
2,137	(242)	Ordinary
22,234	(2,518)	Ordinary
6.437	(729)	Ordinary
29 669	(3.350)	Ordinary
CC) CL	(1 520)	Codings
224(C)	(125,1)	Collialy
1,041	(435)	Ordinary
19,373	(2,194)	Ordinary
3,841	(435)	Ordinary
52,068	(2,839)	Ordinary
2,755	(312)	Ordinary
2,137	(242)	Ordinary
15.929	(1.804)	Ordinary
16 774	(1 843)	Ordinary
30.450	(500,1)	Supplied of
501.02	(, 5, 2)	Clourary
10,181	(1,133)	Ordinary
/89'5	(644)	Ordinary
6,313	(715)	Ordinary
14,896	(1,687)	Ordinary
777	(88)	Ordinary
977.77	(3.146)	Ordinary
C11/17	(040.0)	Ordinary Columbia
015,62	(5,889)	Ordinary
17,219	(1,950)	Ordinary
13,704	(1,552)	Ordinary
6,146	(969)	Ordinary
9 042	(1 024)	Ordinary
010.01	(1 305)	i i i
15,2,1	(000,1)	S G
159,0	(15/)	Ordinary
14,499	(1,642)	Ordinary
2,234	(253)	Ordinary
1 554	(176)	Ordinary
3 532	(400)	Ordinan
2272	(1 707)	Ordinary.
6//'61	(1,/6/,	Ordinary
6	(1)	Ordinary
24,618	(2,788)	Ordinary
31,611	(3,580)	Ordinary
5.113	(623)	Ordinary
8.441	(956)	Ordinary
305.01	(1 207)	Ordinan
12,330	(1,65,1)	Oldinary
107,8	(1,042)	Ordinary
1,280	(145)	Ordinary
38 764	(05. 7)	Ordinary
to / 00	(4) JAC	Ordinary
	2.2.73 2.2.23 2.2.24 2.2.24 13.422 13.422 13.422 13.423 15.29 15.29 15.29 15.29 15.29 15.29 15.29 15.29 15.29 16.1489 17.7 25.510 17.79	

29-Apr-14 29-Apr-14	UBS Securities Australia Ltd				
41-1dw-22	I IDC Commission Assessed and	Buy	1,585,089	177,595	Ordinary
	Upo comilies Australia Liu	- A	866,87	(3,250)	Ordinary
43-451-14	Obs securities Australia Ltd	5uy	106,386	43,000	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Sell	385,409	(43,000)	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	325,100	36,396	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	3,552	400	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	100	Options
29-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (A/N	200	Options
29-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative C	Ϋ́	287	Options
29-Apr-14	UBS Securities Australia I td	Enter Into Derivative (4//0	414	Options
29. Apr. 1.4	I IBC Global Agest Management	200	163.606	100.01	
20 457 14	LIOS MACANE ASSEL MAINAGEMENT	puy C-11	060,201	18,295	Orginary
20 4-1-4	ODS Wealth Management Australia Ltd	Sell	13,408	(606,1)	Ordinary
23-Apr-14	UBS Wealth Management Australia Ltd	Buy	2,491	5/6	Ordinary
29-Apr-14	UBS Wealth Management Australia Ltd	Buy	11,207	1,255	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	2,045	230	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	4,072	458	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Sell	51,615	(2,806)	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	14,321	1,610	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	36	4	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	18	2	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	4,463	205	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	2,098	236	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	4,187	471	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	53	9	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buv	400	45	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buv	1 574	1771	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buv	3 197	359	Ordinary
29-Apr-14	UBS Securities Australia I1d	Buy	5 352	603	Grain
29-Apr-14	I IRC Securities Australia I td	P _{1.1V}	222,0	60	Ordinary
29-Apr-14	LIBS Securities Australia 1td	Risk	1 262	147	ordinary.
29-Anr-14	I IBC Cocurities Australia Ltd	Biox	080	7 00	Ordinary
20 /05	LIBC Contribute Australia 144	Duy	500	200	Oldinary
29 Apr. 14	LIBS Committee Australia Ltd	Day Bury	72	n	Ordinary
20 00 14	Tibe committee Australia Ltd	puy Duri	100,0	200	or distary
29-Apr-14	UBS Securities Australia Ltd	Suy	580	//	Ordinary
25 Apr - 14	Obs Securities Australia Lia	âna	904	40	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	711	08	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	2,160	243	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	21,025	2,365	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	18	2	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Buy	2,311	260	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Sell	3,764	(418)	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Sell	3,780	(420)	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Sell	2.090	(020)	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Sell	2.090	(232)	Ordinary
29-Anr-14	I BS Securities Australia Ltd	Sell	1045	(116)	Ordinary
29-Apr-14	I IRC Cocucitios Australia I to	100	1045	(110)	O'dillay.
20 Apr 14	Post Securities Australia Ltd	Jen Jen	040,1	(110)	Calmary
20.00	Por Commercial Commerc	100	0,0,5	(400)	Commany
29-Apr-14	USC Societion Australia 144	nec S	060	(4)	Ordinary
20 00 12	1 100 Commission Australia Ltd	Jell Jell	620,0	(000)	Ordinary
20 Apr 24	Obs Securities Australia Liu	lac l	21,075	(2,342)	Ordinary
29-Apr-14	UBS Securities Australia Ltd	Sell	/88//7	(3,093)	Ordinary
20 A = 14	Obs securities Australia Ltd	iac S	143,550	(056,51)	Ordinary
29-Apr-14	UBS Securities Australia Ltd	lie C	48,915	(5,435)	Ordinary
30-Apr-14	UBS AG Australia Branch	Sell	39,336	(4,414)	Ordinary
30-Apr-14	UBS AG AUstralia Branch	les l	502,883	(8/5,95)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	5,484	(615)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Buy	141,629	15,900	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Je J	4,472	(498)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	43,419	(4,875)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Buy	194,330	21,790	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Buy	130,593	14,638	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	110,880	(12,427)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative C	AN	(200)	Options
30-Apr-14	UBS Wealth Management Australia Ltd	Buy	13,776	1,548	Ordinary
30-Apr-14	UBS Wealth Management Australia Ltd	ES.	14,074	(1.582)	Ordinary
	מיים היים וויים וויים ומלווים וויים וויים ביים	100	11011	11001	5

1615 Scorning Autorials Ltd 5611 1513 1513 1513 1615 Scorning Autorials Ltd 5611 1513 151	24-Apr-14	UBS Securities Australia Ltd	Sell	1 467	(166)	Ordinary
BIS Securities Autorials Ltd. Sell 11,231 (1,404) BIS Securities Autorials Ltd. Sell 21,232 (1,205) BIS Securities Autorials Ltd. Sell 31,232 (1,205) BIS Securities Autorials Ltd. Sell 31,232 (1,205) BIS Securities Autorials Ltd. Sell 31,232 (1,205) BIS	4-Apr-14	UBS Securities Australia Ltd	Sell	3,032	(343)	Ordinary
USS Securities Australia Ltd Sell 2315 C425 C450	4-Apr-14	UBS Securities Australia Ltd	ll-85	12,411	(1,404)	Ordinary
USS Securities Australia Ltd Sell 422 423 (450 428 424 4	4-Apr-14	UBS Securities Australia Ltd	Sell	3,315	(375)	Ordinary
USS Securities Australia Ltd Self 1,641 1,642 1,590	4-Apr-14	UBS Securities Australia Ltd	ll Sell	7,320	(828)	Ordinary
USS Securities Australia Ltd Self	4-Apr-14	UBS Securities Australia Ltd	Sell	442	(20)	Ordinary
USS Securities Australia Ltd Sell 1,359 (440) USS Securities Australia Ltd Sell 1,350 (591) (198) (198) (198) (199)	4-Apr-14	UBS Securities Australia Ltd	Self	1,681	(190)	Ordinary
USS scorringes Australia LLIC Sell 12,722 (1,386) USS scorringes Australia LLIC Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell 1,549 (1,594) USS scorringes Australia LLIC Sell Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell Sell Sell 1,548 (1,594) USS scorringes Australia LLIC Sell Sell Sell Sell 1,544 (1,594) USS scorringes Australia LLIC Sell	4-Apr-14	UBS Securities Australia Ltd	⊮es	3,892	(440)	Ordinary
UBS Securities Australia Ltd Sell Sell 4,933 (4)9 UBS Securities Australia Ltd Sell Sell 6,500 (750) UBS Securities Australia Ltd Sell Sell 1,3260 (700) UBS Securities Australia Ltd Sell Sell Sell 1,3260 (700) UBS Securities Australia Ltd Sell Sell Sell 1,3260 (700) UBS Securities Australia Ltd Sell Sell Sell 1,3260 (700) UBS Securities Australia Ltd Sell Sell Sell 1,3260 (700) UBS Securities Australia Ltd Sell Sell Sell 1,3200 (700) UBS Securities Australia Ltd Sell Sell Sell 1,3200 (700) UBS Securities Australia Ltd Sell Sell Sell Sell Sell Sell Sell Sell (700) UBS Securities Australia Ltd Sell Sel	4-Apr-14	UBS Securities Australia Ltd	38	12,252	(1,386)	Ordinary
USS Securities Australia Ltd Seil 6500 (150) USS Securities Australia Ltd Seil 6500 (150) (1	4-Apr-14	UBS Securities Australia Ltd	Sell l	433	(48)	Ordinary
UBS Securities Australia Ltd Sell 18.00 (705)	4-Apr-14	UBS Securities Australia Ltd	Sell	4,508	(510)	Ordinary
USS Securities Australia Ltd Sell Sell CO0946 CO0946 CO095 CO0946 CO0946 CO095 CO0946 CO0	4-Apr-14	UBS Securities Australia Ltd	Sell	0.69.9	(220)	Ordinary
UBS Securities Australia Ltd Sell Sell B 722 (994) UBS Securities Australia Ltd Sell Sell B 722 (1914) UBS Securities Australia Ltd Sell Sell B 203 (1,500) UBS Securities Australia Ltd Sell Sell B 203 (1,500) UBS Securities Australia Ltd Sell Sell B 203 (1,500) UBS Securities Australia Ltd Sell S	4-Apr-14	UBS Securities Australia Ltd	Sell	1.848	(209)	Ordinary
Libit Securities Australia Ltd Sell 1,929 (1,900) UBS Securities Australia Ltd Sell 204 (1,300) UBS Securities Australia Ltd Sell 204 (1,300) UBS Securities Australia Ltd Sell 21,741 (2,458) UBS Securities Australia Ltd Sell 21,741 (2,458) UBS Securities Australia Ltd Sell 1,14,99 (1,300) UBS Securities Australia Ltd Sell 1,234 (2,64) UBS Securities Australia Ltd Sell 1,322 (1,300) UBS Securities Australia Ltd Sell 1,322 (1,00) UBS Securities Australia Ltd Sell 1,322 (1,00) UBS Securities Australia Ltd Sell 2,170 (1,00) UBS Securities Australia Ltd Sell 3,22 (1,00) UBS Securities Australia Ltd Sell 3,24 (2,46) UBS Securities Australia Ltd Sell 8,24 (1,00) UBS Securities Australia Ltd Sell 8,24 (1,00) UBS Secu	4-Apr-14	UBS Securities Australia Ltd	Sell	8,792	(984)	Ordinary
UBS Securities Australia Ltd Sell 13,260 (1,500) UBS Securities Australia Ltd Sell 27 (23) UBS Securities Australia Ltd Sell 27 (101) UBS Securities Australia Ltd Sell 2,34 (2,458) UBS Securities Australia Ltd Sell 1,585 (1,733) UBS Securities Australia Ltd Sell 1,586 (1,733) UBS Securities Australia Ltd Sell 1,586 (1,734) UBS Securities Australia Ltd Sell 1,586 (1,732) UBS Securities Australia Ltd Sell 1,514 (1,739) UBS Securities Australia Ltd Sell 1,514 (1,730) UBS Securities Australia Ltd Sell 1,446 (1,60) UBS Securities Australia Ltd Sell 2,710 (2,40) UBS Securities Australia Ltd Sell 2,70 (1,60) UBS Securities Australia Ltd Sell 2,70 (2,40) UBS Securities Australia Ltd Sell 2,70 (1,60) UBS Securities Au	4-Apr-14	UBS Securities Australia Ltd	Sell	8.792	(994)	Ordinary
USS Securities Australia Ltd Sell 203 (23) USS Securities Australia Ltd Sell 203 (24) USS Securities Australia Ltd Sell 2,741 (2,458) USS Securities Australia Ltd Sell 8,523 (1,459) USS Securities Australia Ltd Sell 1,548 (1,793) USS Securities Australia Ltd Sell 1,548 (1,793) USS Securities Australia Ltd Sell 1,923 (1,793) USS Securities Australia Ltd Sell 1,923 (1,701) USS Securities Australia Ltd Sell 1,923 (1,701) USS Securities Australia Ltd Sell 1,923 (1,701) USS Securities Australia Ltd Sell 2,170 (1,401) USS Securities Australia Ltd Sell 3,170 (1,401) USS Securities Australia Ltd Sell 3,170 (1,401) USS Securities Australia Ltd Sell 3,170 (1,401) USS Securities Australia Ltd Sell 8,170 (1,401) USS Secu	4-Anr-14	UBS Securities Australia I td	وا	13 260	(1500)	Ordinary
UBS Securities Australia Ltd Sell 6.0 UBS Securities Australia Ltd Sell 2.7 UBS Securities Australia Ltd Sell 2.7 UBS Securities Australia Ltd Sell 8.9 UBS Securities Australia Ltd Sell 5.83 UBS Securities Australia Ltd Sell 1.546 UBS Securities Australia Ltd Sell 1.534 UBS Securities Australia Ltd Sell 1.534 UBS Securities Australia Ltd Sell 1.546 UBS Securities Australia Ltd Sell 1.546 UBS Securities Australia Ltd Sell 1.546 UBS Securities Australia Ltd Sell 2.170 UBS Securities Australia Ltd Sell 6.42 UBS Securities Australia Ltd Sell 6.42 UBS Securities Australia Ltd Sell <td>1-Anr-14</td> <td>I BS Securities Australia I to</td> <td>100</td> <td>500</td> <td>(50)</td> <td>Ordinany</td>	1-Anr-14	I BS Securities Australia I to	100	500	(50)	Ordinany
UBS Securities Australia Ltd Sell 27.7 (37) UBS Securities Australia Ltd Sell 27.3 (101) UBS Securities Australia Ltd Sell 15.629 (101) UBS Securities Australia Ltd Sell 1.549 (101) UBS Securities Australia Ltd Sell 1.749 (101) UBS Securities Australia Ltd Sell 1.523 (101) UBS Securities Australia Ltd Sell 2.746 (100) UBS Securities Australia Ltd Sell 8.7756 (100) UBS Securities Australia Ltd<	A. Apr. 1.4	(BC Constitute Australia 14d	130	600	(50)	Ordinary Ordinary
UBS Securities Australia Ltd Sell 21,41 (2,458) UBS Securities Australia Ltd Sell 15,859 (1,793) UBS Securities Australia Ltd Sell 5,334 (1,793) UBS Securities Australia Ltd Sell 1,5859 (1,793) UBS Securities Australia Ltd Sell 1,594 (1,700) UBS Securities Australia Ltd Sell 1,914 (2,710) UBS Securities Australia Ltd Sell 1,914 (2,710) UBS Securities Australia Ltd Sell 1,914 (2,710) UBS Securities Australia Ltd Sell 1,924 (2,710) UBS Securities Australia Ltd Sell 1,924 (77) UBS Securities Australia Ltd Sell 2,170 (1440) UBS Securities Australia Ltd Sell 2,170 (1440) UBS Securities Australia Ltd Sell 87,600 (150) UBS Securities Australia Ltd Sell 87,600 (150) UBS Securities Australia Ltd Sell 87,600 (150) UBS		Upo consider Australia Lid	liac o	27	(15)	Oldinary
UBS Securities Australia Ltd Sell 21,741 (4,729) UBS Securities Australia Ltd Sell 1,741 (4,60) UBS Securities Australia Ltd Sell 1,534 (1,60) UBS Securities Australia Ltd Sell 1,734 (1,60) UBS Securities Australia Ltd Sell 1,734 (1,500) UBS Securities Australia Ltd Sell 1,734 (1,500) UBS Securities Australia Ltd Sell 1,933 (1,500) UBS Securities Australia Ltd Sell 1,933 (1,500) UBS Securities Australia Ltd Sell 2,770 (24,500) UBS Securities Australia Ltd Sell 2,770 (24,600) UBS Securities Australia Ltd Sell 2,770 (24,600) UBS Securities Australia Ltd Sell 2,700 (1,00) UBS Securities Australia Ltd Sell 3,810 (1,00) UBS Securities Australia Ltd Sell 1,700 (24,50) UBS Securities Australia Ltd Buy 2,7,60 (1,00)	A Pillar	DDS Securities Australia Liu	lac J	/7	(5)	Ordinary
UBS Securities Australia Ltd Sell 15.89 (10) UBS Securities Australia Ltd Sell 15.89 (17.93) UBS Securities Australia Ltd Sell 2.334 (2.64) UBS Securities Australia Ltd Sell 11.499 (1.793) UBS Securities Australia Ltd Sell 11.499 (1.700) UBS Securities Australia Ltd Sell 11.499 (1.700) UBS Securities Australia Ltd Sell 1.914 (2.71) UBS Securities Australia Ltd Sell 1.924 (1.700) UBS Securities Australia Ltd Sell 2.81 (1.000) UBS Securities Australia Ltd Sell 2.170 (1.64) UBS Securities Australia Ltd Sell 2.340 (1.000) UBS Securities Australia Ltd Sell 3.24 cor. (1.000) UBS Securities Australia Ltd Sell 3.24 cor. (1.000) UBS Securities Australia Ltd Sell 8.80 cor. (1.000) UBS Securities Australia Ltd Sell 3.24 cor. (1.000) <tr< td=""><td></td><td>Opp parallities Australia Liu</td><td>100</td><td>14/17</td><td>(2,450)</td><td>Ordinary</td></tr<>		Opp parallities Australia Liu	100	14/17	(2,450)	Ordinary
UBS Securities Australia Ltd Sell 15,859 (1,733) UBS Securities Australia Ltd Sell 15,854 (2,100) UBS Securities Australia Ltd Sell 11,493 (1,000) UBS Securities Australia Ltd Sell 1,923 (2,100) UBS Securities Australia Ltd Sell 1,923 (1,100) UBS Securities Australia Ltd Sell 2,100 (2,460) UBS Securities Australia Ltd Sell 2,100 (2,460) UBS Securities Australia Ltd Sell 3,801 (4,40) UBS Securities Australia Ltd Sell 3,801 (4,40) UBS Securities Australia Ltd Sell 3,801 (4,00) UBS Securities Australia Ltd Sell 3,801 (4,00) UBS Securities Australia Ltd Sell 3,801 (4,00)	1	UBS Securities Australia Litu	- N	660	(101)	Orginary
UBS Securities Australia Ltd Sell 1,5,859 (1,739) UBS Securities Australia Ltd Sell 1,3,859 (1,730) UBS Securities Australia Ltd Sell 1,3,689 (1,730) UBS Securities Australia Ltd Sell 1,914 (2,100) UBS Securities Australia Ltd Sell 1,914 (2,10) UBS Securities Australia Ltd Sell 1,946 (1,00) UBS Securities Australia Ltd Sell 1,946 (1,00) UBS Securities Australia Ltd Sell 1,446 (1,64) UBS Securities Australia Ltd Sell 2,170 (1,64) UBS Securities Australia Ltd Sell 3,881 (1,64) UBS Securities Australia Ltd Sell 3,881 (1,40) UBS Securities Australia Ltd Sell 3,881 (4,40) UBS Securities Australia Ltd Sell 3,93,229 (4,00) UBS Securities Australia Ltd Sell 8,045 (1,00) UBS Securities Australia Ltd Buy 33,400 (1,00)	Т	OBS Securities Australia Liu	i k	040	(ia)	Ordinary
UBS Securities Australia Ltd Sell 1,334 (1300) UBS Securities Australia Ltd Sell 11,499 (1,300) UBS Securities Australia Ltd Sell 1,524 (2,100) UBS Securities Australia Ltd Sell 1,523 (2,100) UBS Securities Australia Ltd Sell 1,523 (100) UBS Securities Australia Ltd Sell 1,923 (2,10) UBS Securities Australia Ltd Sell 1,923 (2,10) UBS Securities Australia Ltd Sell 1,446 (16,4) UBS Securities Australia Ltd Sell 3,881 (4,40) UBS Securities Australia Ltd Sell 3,881 (440) UBS Securities Australia Ltd Sell 3,881 (440) UBS Securities Australia Ltd Sell 3,881 (440) UBS Securities Australia Ltd Sell 87,600 3,600 UBS Securities Australia Ltd Sell 87,600 3,600 UBS Securities Australia Ltd Sell 87,700 (5,440) UBS Securit	- [UBS Securities Australia Ltd	Ne l	15,859	(1,793)	Ordinary
UBS Securities Australia Ltd Sell 11,499 (1,300) UBS Securities Australia Ltd Sell 1,914 (2,100) UBS Securities Australia Ltd Sell 1,914 (2,17) UBS Securities Australia Ltd Sell 1,914 (2,17) UBS Securities Australia Ltd Sell 1,923 (100) UBS Securities Australia Ltd Sell 1,923 (110) UBS Securities Australia Ltd Sell 1,924 (164) UBS Securities Australia Ltd Sell 2,170 (165) UBS Securities Australia Ltd Sell 2,170 (165) UBS Securities Australia Ltd Sell 3,81 (401) UBS Securities Australia Ltd Sell 3,82 (100) UBS Securities Australia Ltd Sell 3,82 (100) UBS Securities Australia Ltd Sell 3,93 (100) UBS Securities Australia Ltd Buy 3,93 (100) UBS Securities Australia Ltd Buy 3,17,485 3,000 UBS Securities Australia	4-Apr-14	UBS Securities Australia Ltd	Seli	2,334	(264)	Ordinary
UBS Securities Australia Ltd Seil 18,564 (2,100) UBS Securities Australia Ltd Seil 1,914 (217) UBS Securities Australia Ltd Seil 1,924 (7,10) UBS Securities Australia Ltd Seil 1,923 (7,10) UBS Securities Australia Ltd Seil 1,923 (7,10) UBS Securities Australia Ltd Seil 2,170 (246) UBS Securities Australia Ltd Seil 2,170 (246) UBS Securities Australia Ltd Seil 2,170 (246) UBS Securities Australia Ltd Seil 3,881 (440) UBS Securities Australia Ltd Seil 3,881 (440) UBS Securities Australia Ltd Seil 87,656 (32) UBS Securities Australia Ltd Seil 87,656 (43,27) UBS Securities Australia Ltd Buy 37,656 (43,27) UBS Securities Australia Ltd Buy 37,657 (100,000) UBS Securities Australia Ltd Buy 37,657 (100,000) UBS Secur	1-Apr-14	UBS Securities Australia Ltd	Sell	11,499	(1,300)	Ordinary
UBS Securities Australia Ltd Sell 1,914 (1) UBS Securities Australia Ltd Sell 1,914 (217) UBS Securities Australia Ltd Sell 627 (70) UBS Securities Australia Ltd Sell 1,923 (18) UBS Securities Australia Ltd Sell 1,923 (18) UBS Securities Australia Ltd Sell 1,923 (18) UBS Securities Australia Ltd Sell 2,170 (246) UBS Securities Australia Ltd Sell 2,170 (246) UBS Securities Australia Ltd Sell 3,881 (440) UBS Securities Australia Ltd Sell 3,881 (440) UBS Securities Australia Ltd Sell 8,93,20 (4,732) UBS Securities Australia Ltd Sell 8,93,20 (4,732) UBS Securities Australia Ltd Buy 31,405 (3,40) UBS Securities Australia Ltd Buy 31,405 (4,732) UBS Securities Australia Ltd Buy 31,405 (4,10) UBS Securities Australia	1-Apr-14	UBS Securities Australia Ltd	Sell	18,564	(2,100)	Ordinary
UBS Securities Australia Ltd Seli 1,914 (217) UBS Securities Australia Ltd Sell 60) (60) UBS Securities Australia Ltd Sell 62 (70) UBS Securities Australia Ltd Sell 1,923 (10) UBS Securities Australia Ltd Sell 1,923 (10) UBS Securities Australia Ltd Sell 2,170 (246) UBS Securities Australia Ltd Sell 3,881 (40) UBS Securities Australia Ltd Sell 9 (11) UBS Securities Australia Ltd Sell 8,76,50 (3,370) UBS Securities Australia Ltd Buy 324,00 36,700 UBS Securities Australia Ltd Buy 37,00 36,700 UBS Securities Australia Ltd Buy 76,780 (43,722) UBS Securities Australia Branch <td>1-Apr-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>6</td> <td>(1)</td> <td>Ordinary</td>	1-Apr-14	UBS Securities Australia Ltd	Sell	6	(1)	Ordinary
UBS Securities Australia Ltd Sell EG2 (107) UBS Securities Australia Ltd Sell 1.923 (100) UBS Securities Australia Ltd Sell 1.923 (101) UBS Securities Australia Ltd Sell 2.170 (246) UBS Securities Australia Ltd Sell 2.170 (246) UBS Securities Australia Ltd Sell 2.170 (246) UBS Securities Australia Ltd Sell 3.81 (440) UBS Securities Australia Ltd Sell 3.81 (440) UBS Securities Australia Ltd Sell 3.81 (440) UBS Securities Australia Branch Sell 8.72 (35) UBS Securities Australia Branch Sell 8.72 (43) UBS Securities Australia Branch Sell 8.76 (44.732) UBS Securities Australia Branch Sell 8.76 (43.705) UBS Securities Australia Branch Sell 8.76 (43.705) UBS Securities Australia Branch Buy 2.70 (5.416) UBS Securitie	-Apr-14	UBS Securities Australia Ltd	Sell	1,914	(217)	Ordinary
UBS Securities Australia Ltd Sell 62 (7) UBS Securities Australia Ltd Sell 1,923 (180) UBS Securities Australia Ltd Sell 1,446 (164) UBS Securities Australia Ltd Sell 2,70 (164) UBS Securities Australia Ltd Sell 2,22 (17) UBS Securities Australia Ltd Sell 4,23 (40) UBS Securities Australia Ltd Sell 3,88 (440) UBS Securities Australia Branch Sell 8,70 (1) UBS Securities Australia Branch Sell 8,70 (4,732) UBS Securities Australia Branch Sell 8,70 (4,732) UBS Securities Australia Branch	1-Apr-14	UBS Securities Australia Ltd	Jesi	525	(09)	Ordinary
UBS Securities Australia Ltd Sell 1,923 (100) UBS Securities Australia Ltd Sell 1,923 (218) UBS Securities Australia Ltd Sell 2,170 (246) UBS Securities Australia Ltd Sell 2,170 (246) UBS Securities Australia Ltd Sell 2,23 (43) UBS Securities Australia Ltd Sell 3,881 (440) UBS Securities Australia Ltd Sell 3,881 (440) UBS Securities Australia Ltd Sell 8,90,000 (43) UBS Securities Australia Ltd Sell 8,700 (44,722) UBS AGA Australia Branch Sell 8,700 (44,722) UBS AGA Australia Branch Sell 8,700 (44,722) UBS AGA Australia Ltd Buy 3,4,607 36,700 UBS AGA Australia Branch Sell 8,700 10,93,700 UBS AGA Australia Branch Sell 8,700 1,4,4,220 UBS Securities Australia Ltd Buy 3,4,607 3,6,700 UBS Securities Australia Bran	1-Apr-14	UBS Securities Australia Ltd	Sell	79	(2)	Ordinary
UBS Securities Australia Ltd Sell 1,923 (218) UBS Securities Australia Ltd Sell 2,170 (246) UBS Securities Australia Ltd Sell 2,170 (246) UBS Securities Australia Ltd Sell 2,82 (15) UBS Securities Australia Ltd Sell 4,92 (17) UBS Securities Australia Ltd Sell 3,881 (440) UBS Securities Australia Branch Sell 878,049 (17) UBS Securities Australia Branch Sell 878,049 (10) UBS AG Australia Branch Sell 878,049 (10) UBS AG Australia Branch Sell 878,049 (10) UBS AG Australia Branch Sell 878,049 (10) UBS AG Australia Branch Buy 324,007 (10) UBS Securities Australia Ltd Buy 324,007 (36,938) UBS Securities Australia Ltd Buy 76,788 (44,732) UBS Securities Australia Ltd Buy 76,738 (34,401) UBS Securities Australia Ltd <td>1-Apr-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>882</td> <td>(100)</td> <td>Ordinary</td>	1-Apr-14	UBS Securities Australia Ltd	Sell	882	(100)	Ordinary
UBS Securities Australia Ltd Sell 1,446 (164) UBS Securities Australia Ltd Sell 2,170 (245) UBS Securities Australia Ltd Sell 2,72 (48) UBS Securities Australia Ltd Sell 3,881 (401) UBS Securities Australia Ltd Sell 3,881 (401) UBS Securities Australia Ltd Sell 3,881 (401) UBS Securities Australia Branch Sell 878,049 (100,000) UBS AG Australia Branch Sell 827,656 (33,878) UBS AG Australia Branch Sell 827,656 (33,878) UBS AG Australia Branch Sell 827,656 (33,878) UBS AG Australia Branch Sell 82,656 (44,722) UBS AG Australia Branch Sell 82,656 (43,878) UBS Securities Australia Ltd Buy 31,485 36,093 UBS Securities Australia Branch Sell 85,566 (43,872) UBS Securities Australia Ltd Buy 18,327 (44,572) UBS Securiti	1-Apr-14	UBS Securities Australia Ltd		1.923	(218)	Ordinary
UBS Securities Australia Ltd Sell 2,170 (246) UBS Securities Australia Ltd Sell 282 (48) UBS Securities Australia Ltd Sell 423 (48) UBS Securities Australia Ltd Sell 381 (440) UBS Securities Australia Ltd Sell 3881 (440) UBS Securities Australia Ltd Sell 878 (710) UBS Securities Australia Branch Sell 878 (730) UBS Securities Australia Branch Buy 334,007 36,700 UBS AG Australia Branch Sell 878,009 (100,000) UBS AG Australia Branch Sell 878,009 (100,000) UBS AG Australia Branch Sell 878,009 (100,000) UBS AG Australia Branch Sell 87,000 (100,000) UBS AG Australia Branch Buy 37,400 (100,000) UBS Securities Australia Ltd Buy 149,322 (44,732) UBS Securities Australia Ltd Buy 76,788 (859) UBS Securities Australia Lt	1-Apr-14	UBS Securities Australia Ltd	38	1,446	(164)	Ordinary
UBS Securities Australia Ltd \$ell 1 UBS Securities Australia Ltd \$ell 428 (32) UBS Securities Australia Ltd \$ell 423 (440) UBS Securities Australia Ltd \$ell 3,881 (440) UBS Securities Australia Ltd \$ell 3,881 (440) UBS Securities Australia Branch \$ell 878,049 (7) UBS AG Australia Branch Buy 34,007 36,000 UBS AG Australia Branch Sell 87,656 (44,732) UBS AG Australia Branch Sell 87,656 (44,732) UBS AG Australia Branch Sell 87,656 (44,732) UBS AG Australia Branch Buy 31,465 36,500 UBS AG Australia Branch Buy 32,400 36,564 UBS Securities Australia Ltd Buy 32,400 36,564 UBS Securities Australia Ltd Buy 76,738 (34,431) UBS Securities Australia Ltd Enter Into Derivative C N/A 34 UBS Securities Australia Branch Buy </td <td>1-Apr-14</td> <td>UBS Securities Australia Ltd</td> <td>3</td> <td>2,170</td> <td>(246)</td> <td>Ordinary</td>	1-Apr-14	UBS Securities Australia Ltd	3	2,170	(246)	Ordinary
UBS Securities Australia Ltd Sell 282 (48) UBS Securities Australia Ltd Sell 3.881 (40) UBS Securities Australia Ltd Sell 3.881 (40) UBS Securities Australia Ltd Sell 3.881 (40) UBS Securities Australia Branch Sell 878,049 (10) UBS AG Australia Branch Sell 879,049 (100,000) UBS AG Australia Branch Sell 827,656 (43,878) UBS Securities Australia Ltd Buy 33,400 (100,000) UBS Securities Australia Ltd Buy 149,322 (44,21) UBS Securities Australia Ltd Buy 149,322 (5,416) UBS Securities Australia Ltd Buy 16,382 (1,249) UBS Securities Au	1-Apr-14	UBS Securities Australia Ltd	38	6	ε	Ordinary
UBS Securities Australia Ltd Sell 423 (48) UBS Securities Australia Ltd Sell 3,881 (40) UBS Securities Australia Ltd Sell 3,881 (40) UBS Securities Australia Branch Sell 878,049 (10,000) UBS AGA Australia Branch Buy 324,007 36,700 UBS AGA Australia Branch Sell 827,656 (44,732) UBS AGA Australia Branch Sell 827,656 (44,732) UBS AGA CAustralia Branch Sell 827,656 (44,732) UBS AGA CAustralia Branch Sell 827,656 (44,732) UBS AGA CAustralia Branch Buy 37,405 (5,460) UBS Securities Australia Ltd Buy 31,485 36,337 UBS Securities Australia Ltd Buy 76,738 (5,46) UBS Securities Australia Ltd Buy 76,738 (5,46) UBS Securities Australia Ltd Buy 76,738 (8,4421) UBS Securities Australia Ltd Buy 76,738 (89) UBS Securities Au	1-Apr-14	UBS Securities Australia Ltd	les.	282	(32)	Ordinary
UBS Securities Australia Ltd Sell 3,861 (440) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 878,049 (10) UBS Securities Australia Branch Sell 878,049 (10,0000) UBS AG Australia Branch Sell 87,656 (93,878) UBS AG Australia Branch Sell 87,656 (93,878) UBS AG Australia Branch Sell 87,656 (93,878) UBS AG Australia Branch Sell 87,656 (43,732) UBS AG Australia Branch Buy 31,465 36,938 UBS Securities Australia Ltd Buy 31,465 36,938 UBS Securities Australia Ltd Buy 143,322 16,937 UBS Securities Australia Ltd Buy 76,738 (34,431) UBS Securities Australia Ltd Enter Into Derivative C NA (259) UBS Securities Australia Ltd Buy 75,738 (34,431) UBS Securities Australia Ltd Buy 75,738 (34,431) UBS	1-Apr-14	UBS Securities Australia Ltd	Sei	423	(48)	Ordinary
UBS Securities Australia Ltd Sell 6,442 (170) UBS Securities Australia Ltd Sell 318 (130) UBS Securities Australia Branch Sell 334,007 36,700 UBS AG Australia Branch Sell 334,007 36,700 UBS AG Australia Branch Sell 875,659 (44,732) UBS AG Australia Branch Sell 875,656 (44,732) UBS AG Australia Branch Sell 875,656 (43,878) UBS AG Australia Branch Sell 875,656 (43,878) UBS Securities Australia Ltd Buy 317,485 36,033 UBS Securities Australia Ltd Buy 47,707 (5,460) UBS Securities Australia Ltd Sell 47,707 (5,416) UBS Securities Australia Ltd Enter Into Derivative NA 39 UBS Securities Australia Ltd Enter Into Derivative NA 34 UBS Securities Australia Ltd Buy 10,506 (359) UBS Securities Australia Ltd Buy 1,544 5,249 <t< td=""><td>1-Anr-14</td><td>LIBS Securities Australia Ltd</td><td>Çeji</td><td>3 881</td><td>(440)</td><td>Ordinary</td></t<>	1-Anr-14	LIBS Securities Australia Ltd	Çeji	3 881	(440)	Ordinary
UBS Securities Australia Ltd Sell 6,442 (730) UBS Securities Australia and UBS Securities Australia Branch Sell 378,049 (100,000) UBS AG Australia Branch Buy 37,000 36,700 100,000 UBS AG Australia Branch Buy 37,600 36,700 UBS AG Australia Branch Sell 37,600 36,700 UBS AG Australia Branch Sell 37,600 36,700 UBS Securities Ltd Buy 37,400 2,654 UBS Securities Australia Ltd Buy 37,400 2,654 UBS Securities Australia Ltd Buy 37,707 36,708 UBS Securities Australia Ltd Buy 76,738 8,746 UBS Securities Australia Ltd Buy 76,738 8,746 UBS Securities Australia Ltd Buy 76,738 8,4421) UBS Securities Australia Ltd Enter Into Derivative C NA 1,249 UBS Securities Australia Ltd Enter Into Derivative C NA 1,249 UBS Securities Australia Ltd Buy 7,568 <	-Anr-14	I IRS Societities Australia 14d	Sall	a	(1)	Ordinary
UBS Securities Australia and Enter Into Derivative Only All All All All All All All All All A	Apr. 14	I IBS Societies Australia 14d	loo loo	6443	(062/	Collingry
UBS AGG Australia Branch Sell 878,049 (100,000) UBS AGG Australia Branch Buy 324,007 36,700 UBS AGG Australia Branch Sell 839,229 36,700 UBS AGG Australia Branch Sell 839,229 36,700 UBS AGG Australia Branch Sell 835,229 36,487 UBS Securities Australia Ltd Buy 37,656 (34,378) UBS Securities Australia Ltd Buy 317,465 36,033 UBS Securities Australia Ltd Buy 149,322 16,337 UBS Securities Australia Ltd Sell 47,707 (5,460) UBS Securities Australia Ltd Sell 76,738 8,705 UBS Securities Australia Ltd Enter Into Derivative C NA 34 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Securities Australia Ltd Buy 10,506 (2,249 UBS Securities Australia Ltd Buy 10,506 (32,44 UBS Securities Australia Ltd Buy 10,506 (35,90	1 1	TIDE Consider Australia Ltd	100	2440	(750)	Quillally Coulinally
UBS AGG Australia Branch Seli 8 F8,149 (100,000) UBS AGG Australia Branch Seli 339,229 (44,732) UBS AGG Australia Branch Seli 334,007 (47,322) UBS Securities Matarial and Buy Buy 23,409 2,654 UBS Securities Australia Ltd Buy 23,409 2,654 UBS Securities Australia Ltd Buy 149,322 16,337 UBS Securities Australia Ltd Buy 149,322 16,337 UBS Securities Australia Ltd Buy 76,788 8,705 UBS Securities Australia Ltd Buy 76,788 8,705 UBS Securities Australia Ltd Enter Into Derivative C NA 34 UBS Securities Australia Ltd Enter Into Derivative C NA 12,249 UBS Securities Australia Ltd Enter Into Derivative C NA 12,249 UBS Securities Australia Ltd Enter Into Derivative C NA 12,249 UBS Global Asset Management Buy 7,568 (859) UBS Gobal Asset Management Buy 7,568 <td< td=""><td>+ Idy-</td><td>UBS SECURITIES AUSTRAINA LICA</td><td>lac o</td><td>318</td><td>(35)</td><td>Ordinary</td></td<>	+ Idy-	UBS SECURITIES AUSTRAINA LICA	lac o	318	(35)	Ordinary
UBS AGG Australia Branch Bluy 35,700 UBS AGG Australia Branch Sell 39,220 (4,722) UBS AGG Australia Branch Sell 827,656 (93,878) UBS Securities LIC Buy 23,400 (5,460) UBS Securities Australia Ltd Buy 317,465 (93,878) UBS Securities Australia Ltd Buy 317,465 36,093 UBS Securities Australia Ltd Buy 149,322 16,933 UBS Securities Australia Ltd Sell 47,707 (5,416) UBS Securities Australia Ltd Buy 149,322 16,387) UBS Securities Australia Ltd Buy 47,707 (5,416) UBS Securities Australia Ltd Enter Into Derivative NA 34 UBS Securities Australia Ltd Enter Into Derivative NA 34 UBS Securities Australia Ltd Buy 7,506 12,249 UBS Securities Australia Ltd Buy 7,506 12,249 UBS Securities Australia Ltd Buy 7,506 12,249 UBS Securities Australia	-Apr-14	UBS AG Australia Branch	lles l	8/8,049	(100,000)	Ordinary
UBS AG Australia Branch Sell 393,229 (44,732) UBS Securities Australia Branch Sell 339,229 (44,732) UBS Securities Australia Ltd Buy 23,409 2,654 UBS Securities Australia Ltd Buy 317,485 36,033 UBS Securities Australia Ltd Buy 149,332 16,337 UBS Securities Australia Ltd Buy 76,738 8,705 UBS Securities Australia Ltd Buy 76,738 8,705 UBS Securities Australia Ltd Sell 832,718 (9,4421) UBS Securities Australia Ltd Enner Into Derivative C NA 34 UBS Securities Australia Ltd Enner Into Derivative C NA 34 UBS Securities Australia Ltd Enter Into Derivative C NA 34 UBS Securities Australia Ltd Enter Into Derivative C NA 12,249 UBS Geourlies Australia Ltd Buy 1,568 (859) UBS Securities Australia Ltd Buy 1,584 (17,284) UBS Wealth Management Buy 1,588 (17,2	-Apr-14	UBS AG Australia Branch	Buy	324,007	36,700	Ordinary
UBS Act Australia Branch Sell 827,656 (93,878) UBS Securities LLC Transfer-out NA (5,460) UBS Securities Australia Ltd Buy 31,485 36,093 UBS Securities Australia Ltd Buy 31,485 36,093 UBS Securities Australia Ltd Buy 144,322 16,337 UBS Securities Australia Ltd Sell 47,707 (5,416) UBS Securities Australia Ltd Buy 76,738 8,705 UBS Securities Australia Ltd Enter Into Derivative C NA 34 UBS Securities Australia Ltd Enter Into Derivative C NA 34 UBS Securities Australia Ltd Enter Into Derivative C NA 34 UBS Gobal Asset Management Buy 1,2,49 1,2,49 UBS Securities Australia Ltd Buy 1,688 1,589 UBS Securities Australia Ltd Buy 1,588 1,589 UBS Gobal Asset Management Buy 2,744 2,796 UBS Securities Australia Enanch Buy 2,756 2,776	Apr-14	UBS AG Australia Branch	Sell	393,229	(44,732)	Ordinary
UBS Securities Australia Ltd Figure 1.0. Institute 1.0. Institute 1.0. (5,460) UBS Securities Australia Ltd Buy 317,485 36,033 UBS Securities Australia Ltd Sell 66,566 (9,837) UBS Securities Australia Ltd Buy 143,322 16,537 UBS Securities Australia Ltd Buy 77,738 8,705 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Global Asset Management Buy 51,444 5,844 UBS Global Asset Management Buy 51,444 5,844 UBS Wealth Management Australia Ltd Buy 7,568 (859) UBS Acturities Australia Branch Buy 10,333 1,111 UBS Acturities Australia Branch Buy 221,236 24,776 UBS Acturities Australia Branch	-Apr-14	UBS AG Australia Branch	Seti	827,656	(93,878)	Ordinary
UBS Securities Australia Ltd Buy 32,409 2,654 UBS Securities Australia Ltd Bluy 317,485 36,033 UBS Securities Australia Ltd Buy 149,322 16,337 UBS Securities Australia Ltd Buy 76,738 8,705 UBS Securities Australia Ltd Sell 76,738 8,705 UBS Securities Australia Ltd Sell 76,738 8,4421) UBS Securities Australia Ltd Enter Into Derivative C NA 34 UBS Securities Australia Ltd Enter Into Derivative C NA 12,249 UBS Georgia Australia Ltd Enter Into Derivative C NA 12,249 UBS Global Asset Management Buy 5,144 5,844 UBS Global Asset Management Australia Ltd Sell 1,599 1,193 UBS Global Asset Management Australia Ltd Buy 5,144 5,844 UBS Global Asset Management Australia Enter Buy 1,599 1,193 UBS Gord Australia Enterch Buy 1,544 5,844 UBS Act Australia Branch Buy 1,24,24 <td> Apr-14</td> <td>UBS Securities LLC</td> <td>Transfer-out</td> <td>ΑN</td> <td>(5,460)</td> <td>ADR</td>	Apr-14	UBS Securities LLC	Transfer-out	ΑN	(5,460)	ADR
UBS Securities Australia Ltd Buy 317,485 36,093 UBS Securities Australia Ltd Sell 65,566 (9,837) UBS Securities Australia Ltd Sell 47,707 (5,416) UBS Securities Australia Ltd Buy 77,707 (5,416) UBS Securities Australia Ltd Sell 47,707 (5,416) UBS Securities Australia Ltd Enter Into Derivative NA 34 UBS Securities Australia Ltd Enter Into Derivative NA 34 UBS Securities Australia Ltd Enter Into Derivative NA 34 UBS Securities Australia Ltd Buy 7,568 (859) UBS Global Asset Management Buy 7,568 (859) UBS Securities Australia Ltd Buy 7,568 (859) UBS Securities Australia Ltd Buy 7,568 (859) UBS Securities Australia Branch Buy 21,234 (17,284) UBS AG Australia Branch Buy 21,248 (17,284) UBS AG Australia Branch Sell 154,549 (17,284)	-Apr-14	UBS Securities Australia Ltd	Buy	23,409	2,654	Ordinary
UBS Securities Australia Ltd Self 86,566 (9,837) UBS Securities Australia Ltd Buy 14,332 16,937 UBS Securities Australia Ltd Buy 75,738 8,705 UBS Securities Australia Ltd Self 82,718 (9,421) UBS Securities Australia Ltd Enter Into Derivative NA 34 UBS Securities Australia Ltd Enter Into Derivative NA 39 UBS Securities Australia Ltd Enter Into Derivative NA 39 UBS Georgines Australia Ltd Enter Into Derivative NA 39 UBS Georgines Australia Ltd Buy 51,444 5,844 UBS Georgines Australia Ltd Buy 51,444 5,844 UBS Wealth Management Buy 1,568 (859) UBS Wealth Management Australia Ltd Buy 1,588 1,181 UBS Securities Australia Branch Buy 10,306 1,178 UBS Ad Australia Branch Buy 10,306 1,17,284 UBS Ad Australia Branch Sell 1,27,366 39,567	-Apr-14	UBS Securities Australia Ltd	Buv	317.485	36.093	Ordinary
UBS Securities Australia Ltd Buy 149;322 16,937 UBS Securities Australia Ltd Sell 76,738 (5,416) UBS Securities Australia Ltd Sell 76,738 (5,416) UBS Securities Australia Ltd Enter Into Derivative C NA 34 UBS Securities Australia Ltd Enter Into Derivative C NAA (299) UBS Securities Australia Ltd Enter Into Derivative C NAA (299) UBS Securities Australia Ltd Enter Into Derivative C NAA (299) UBS Global Asset Management Buy 51,444 15,249 UBS Securities Australia Ltd Buy 7,568 (859) UBS Securities Australia Ltd Buy 221,236 (877) UBS AGG Australia Branch Buy 221,236 24,776 UBS AGG Australia Branch Sell 154,549 (17,284) UBS AGG Australia Branch Sell 270,788 (93,937) UBS AGG Australia Branch Sell 270,788 (93,537) UBS AGG Australia Branch Sell 1,535,600 <	-Apr-14	UBS Securities Australia I td	Sel	86 566	(9.837)	Ordinary
UBS Securities Australia Ltd Sell 47,707 (5,416) UBS Securities Australia Ltd Buy 76,738 8,705 UBS Securities Australia Ltd Sell 76,738 8,705 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Global Asset Management Buy 107,906 12,249 UBS Global Asset Management Buy 51,444 5,844 UBS Global Asset Management Buy 10,393 1,83 UBS Govarities Australia Ltd Buy 10,393 1,83 UBS Securities Australia Branch Buy 221,236 24,776 UBS AG Australia Branch Sell 270,788 (17,284)	-Apr-14	UBS Sequrities Australia Itd	Bliv	149 322	16 937	Ordinary
UBS Securities Australia Ltd Buy 76,738 8,705 UBS Securities Australia Ltd Sell Enter Into Derivative NA 34 UBS Securities Australia Ltd Enter Into Derivative NA 39 UBS Securities Australia Ltd Enter Into Derivative NA 39 UBS Securities Australia Ltd Enter Into Derivative NA 39 UBS Gecurities Australia Ltd Enter Into Derivative NA 39 UBS Gecurities Australia Ltd Buy 51,444 5,249 UBS Wealth Management Buy 7,568 (859) UBS Securities Australia Ltd Buy 1,638 1,181 UBS Securities Australia Branch Buy 10,393 1,181 UBS Ad Australia Branch Sell 112,549 (17,284) UBS Ad Australia Branch Sell 114,545 39,577 UBS Ad Australia Branch Sell 27,736 (17,284) UBS Ad Australia Branch Sell 1164,549 (17,284) UBS Ad Cuordon Branch Sell 1,573,690 (17,	-Anr-14	I IBS Securities Australia 14d	Coll	707.74	(5,716)	Coding
UBS Securities Australia Ltd Puly Puly Puly Puls Puls		DO COMPLETE ACTION OF THE LEAD	100	11/10/	(014°C)	Oldinary
UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Securities Australia Ltd Enter Into Derivative C NA 39 UBS Global Asset Management Buy 107,906 12,249 UBS Global Asset Management Buy 51,444 5,844 UBS Wealth Management Australia Ltd Buy 10,393 1,93 UBS Activation and Asset Management Australia Ltd Buy 221,236 (859) UBS Act Australia Branch Buy 221,236 24,776 UBS Act Australia Branch Sell 270,788 (93,987) UBS Act Australia Branch Sell 270,788 (93,987) UBS Act Australia Branch Sell 270,788 (17,284) UBS Act Australia Branch Sell 270,788 (17,284) UBS Act Australia Branch Sell 270,788 (10,408) UBS Act Australia Branch Sell 20,478 (17,284)	- HDI - 14	UBS Securities Australia Ltd	gny.	86/'9/	3,705	Ordinary
UBS Securities Australia Ltd Enter into Derivative (LBS Securities Australia Ltd Enter into Derivative (LBS Securities Australia Ltd Enter into Derivative (LBS Securities Australia Ltd NA 39 UBS Securities Australia Ltd Enter into Derivative (LBS Securities Australia Ltd Enter into Derivative (LBS Securities Australia Ltd 10.7906 12.249 UBS Wealth Management Australia Ltd Sell 7,568 (893) UBS Securities Australia Ltd Buy 1,698 193 UBS Securities Australia Enanch Buy 221,236 24,776 UBS AG Australia Branch Sell 115,549 (17,284) UBS AG Australia Branch Buy 221,236 24,776 UBS AG Australia Branch Buy 227,236 33,597 UBS AG Australia Branch Sell 270,789 (17,284) UBS AG Australia Branch Sell 270,789 (17,584) UBS AG Australia Branch Sell 270,789 (17,584)	-Apr-14	Upo Securities Australia Ltd	Sell	81/18	(34,421)	Ordinary
UBS Securities Australia Ltd Enter into Derivative NA 39 UBS Securities Australia Ltd Enter into Derivative NA (399) UBS Global Asset Management Buy 10,7906 12,249 UBS Global Asset Management Buy 5,844 (859) UBS Securities Australia Ltd Buy 1,588 1,839 UBS Securities Australia Enach Buy 221,236 24,776 UBS AG Australia Branch Sell 221,236 24,776 UBS AG Australia Branch Buy 220,236 93,587 UBS AG Australia Branch Sell 270,788 (30,408) UBS AG Australia Branch Sell 270,788 (30,408) UBS AG London Branch Sell 1,535,590 (17,294)	8-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	34	Options
UBS Securities Australia End Enter Into Derivative of UBS Securities Australia End INA (299) UBS Global Asset Management Buy 107,906 12,249 UBS Global Asset Management Buy 51,444 5,844 UBS Wealth Management Australia Ltd Buy 7,568 (859) UBS Securities Australia Ltd Buy 10,333 1,811 UBS AG Australia Branch Buy 221,236 24,776 UBS AG Australia Branch Sell 154,549 (17,284) UBS AG Australia Branch Buy 270,788 (93,987) UBS AG Australia Branch Sell 270,788 (93,987) UBS AG Australia Branch Sell 270,788 (93,408) UBS AG London Branch Sell 1,573,690 (177,595)	3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative (Ϋ́	33	Options
UBS Global Asset Management Buy 107,906 12,249 UBS Global Asset Management Buy 51,444 5,444 UBS Wealth Management Australia Ltd Sell 7,568 (859) UBS Securities Australia Ltd Buy 1,688 193 UBS AG Australia Branch Buy 221,236 24,776 UBS AG Australia Branch Sell 154,549 (17,284) UBS AG Australia Branch Buy 221,236 24,776 UBS AG Australia Branch Sell 154,549 (17,284) UBS AG Australia Branch Sell 270,789 (30,408) UBS AG Cundon Branch Sell 1,573,690 (17,595)	3-Apr-14	UBS Securities Australia Ltd	Enter Into Derivative C	N/A	(533)	Options
UBS Good Asset Management Buy 51,444 5,844 UBS Wealth Management Australia Ltd Sell 7,568 (859) UBS Securifies Australia Ltd Buy 1,688 1,181 UBS Ad Australia Branch Buy 221,236 24,776 UBS Ad Australia Branch Sell 15,454 93,977 UBS AG Australia Branch Sell 270,788 (30,408) UBS AG Australia Branch Sell 270,788 (30,408) UBS AG Loudon Branch Sell 1,73,595) (17,595)	3-Apr-14	UBS Global Asset Management	Buy	107,906	12,249	Ordinary
UBS Wealth Management Australia Ltd Sell 7.568 (859) UBS Securities Australia Ltd Buy 1,638 193 UBS Securities Australia Ltd Buy 221,236 24,776 UBS AG Australia Branch Sell 221,236 24,776 UBS AG Australia Branch Sell 154,549 (17,284) UBS AG Australia Branch Sell 270,788 (30,408) UBS AG Lostralia Branch Sell 270,788 (30,408) UBS AG Lordon Branch Sell 1,573,690 (17,295)	8-Apr-14	UBS Global Asset Management	Buy	51,444	5,844	Ordinary
UBS Securities Australia Ltd Buy 1,698 193 UBS Securities Australia Enach Buy 10,393 1,181 UBS AG Australia Branch Buy 221,236 24,776 UBS AG Australia Branch Sell 1154,549 (17,284) UBS AG Australia Branch Buy 839,206 39,597 UBS AG Australia Branch Sell 270,789 (30,408) UBS AG Lordon Branch Sell 1,573,690 (777,595)	8-Apr-14	UBS Wealth Management Australia Ltd	Sell	7,568	(828)	Ordinary
UBS Securities Australia Etanch Buy 10,333 1,181 UBS AG Australia Branch Buy 221,236 24,776 UBS AG Australia Branch Sell 154,549 (17,284) UBS AG Australia Branch Buy 899,206 93,597 UBS AG Australia Branch Sell 270,789 (30,408) UBS AG London Branch Sell 1,573,690 (17,595)	3-Apr-14	UBS Securities Australia Ltd	Buy	1,698	193	Ordinary
UBS AG Australia Branch Buy 221,236 24,776 UBS AG Australia Branch Sell 154,549 (17,284) UBS AG Australia Branch Buy 839,206 93,977 UBS AG Australia Branch Sell 270,788 (30,408) UBS AG London Branch Sell 1,573,690 (177,595)	3-Apr-14	UBS Securities Australia Ltd	Buy	10,393	1,181	Ordinary
UBS AG Australia Branch Sell 1154,549 (17,284) UBS AG Australia Branch Buy 839,206 39,597 UBS AG Australia Branch Sell 270,789 (30,408) UBS AG London Branch Sell 1,573,690 (777,595)	-Apr-14	UBS AG Australia Branch	Buy	221,236	24,776	Ordinary
UBS AG Australia Branch Buy 839,206 93,957 UBS AG Australia Branch Sell 270,789 (30,408) UBS AG London Branch Sell 1,573,690 (177,595)	-Apr-14	UBS AG Australia Branch	Sell	154,549	(17,284)	Ordinary
UBS AG Australia Branch Sell 270,789 (30,408) UBS AG London Branch Sell 1,573,690 (177,595)	-Apr-14	UBS AG Australia Branch	Buy	839,206	93,957	Ordinary
UBS AG London Branch Sell 1,573,690 (177,595)	-Apr-14	UBS AG Australia Branch	Sell	270,789	(30,408)	Ordinary
(creft) control	-Apr-14	UBS AG London Branch	llas	1,573,690	(177,595)	Ordinary
IRS AG London Branch 177 505 177 505	Anr-14	I IRC AG London Branch	Rist	1 585 089	177 595	Ordinary
Contracti (no			,			20.00

30-Apr-14	UBS Securities Australia Ltd	Sell	08	6)	Ordinary
-17	TOO Committee A control of	1,72	0000	1000/	-
Apr 14	Libs countries Australia Ltd	lac -	799'7	(300)	Ordinary
20-701-14 00 4 2 1 4	Upo Committees Australia Ltd	III O	2,5/2	(782)	Ordinary
30-Api-14	UBS Securities Australia Ltd	Sell	050,11	(1,236)	Ordinary
ADI-14	UBS Securities Australia Ltd	Seli	5,024	(295)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	25,926	(2,900)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	2,181	(244)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	54	(9)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	25.926	(2.900)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Seli	2,968	(332)	Ordinary
30-Apr-14	LIBS Securities Australia Ltd	la)	0.4 4 7 0	(200)	Ordingo
30-Apr-14	I IRS Societities Australia I to	Sell	25 026	(000)	Grainary
20 Apr 14	A DC Countries Assemble 144	200	020,02	(2,300)	Cidinary
1	Opp Securities Australia Liu	ii o	44,700	(DDD(c)	Ordinary
50-Apr-14	Obsisecurities Australia Ltd	Seli	17,353	(1,941)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	16,012	(1,791)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	4,577	(512)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	1,296	(145)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	2,700	(302)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	8,940	(1.000)	Ordinary
80-Apr-14	UBS Securities Australia I td	las	388 8	(373)	Ordinato
-Anr-14	I IRC Securities Australia Ltd	100	0090	(000)	Ordings.
30 Apr 14	I IDC Countries Australia 14d	120	20072	(000)	Cicinary
1 1	Opposed titles Australia Ltd	liac li	30,342	(3,394)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	08	6	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	11,184	(1,251)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	13,187	(1,475)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	5,221	(284)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	20,428	(2,285)	Ordinary
-Apr-14	UBS Securities Australia Ltd	روا	198	(41)	Ordinary
30-Apr-14	BS Securities Australia	3	702	(64)	0
20 Apr 14	100 Counting Assessed and		57/ J	(101)	o comany
1	Opp Securities Australia Ltd	ilac i	057'6	(585)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	17,755	(1,986)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	68	(10)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	5.221	(584)	Ordinary
0-Apr-14	I IBS Congrition Australia 14d	la)	2 007	(344)	
30-Apr.14	Lie Commission Assessed and	100	19575	(0++)	Colliary
1 1 1 1	The Contract Action of the Contract of the Con	100	1/0/71	(0,4,0)	Cidinary
1	ODS Securities Australia Ltd	lac	8///0/	(/ [8//)	Ordinary
SU-Apr-14	UBS Securities Australia Ltd	Sell	16,38/	(1,833)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	14,572	(1,630)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	2,441	(273)	Ordinary
0-Apr-14	UBS Securities Australia Ltd	Sell	8.046	(006)	Ordinary
30-Apr-14	I IBS Securities Australia Ltd	No.	8.073	(809)	, acripio
7 7 7 7 7	The Commission Action Co.	===	500	(505)	Cicinary
+H-ICH-	Obs securities Australia LIG) Sell	176	(103)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Seil	54	(9)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	921	(103)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	10,388	(1,162)	Ordinary
10-Apr-14	UBS Securities Australia Ltd	Sell	921	(103)	Ordinary
30-Apr-14	I IRS Securities Australia I td	===	12 272	(1 304)	Ordinary Prolings
V Apr 14	100 Commission Americal and	====	50,00	(+00'1)	O GILIAI
1	ODS SECULINES AUSTRAINA LIQ	liac liac	12,194	(1,304)	Orginary
SU-Apr-14	UBS Securities Australia Ltd	Sell	8,547	(926)	Ordinary
-Apr-14	UBS Securities Australia ttd	Sell	313	(32)	Ordinary
0-Apr-14	UBS Securities Australia Ltd	Sell	4,470	(200)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Ж	2.968	(332)	Ordinary
-Anr-14	I IBS Sequities Australia 11d	=9	30 306	(3 400)	Ordinan
	DO COMPANY OF THE PARTY OF THE		00000	(001,0)	diam's
30-Apr-14	UBS securities Australia Ltd	Sell	14,5/2	(1,630)	Ordinary
-Apr-14	UBS Securities Australia Ltd	Sell	29,1001	(3,255)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	- Se	11,264	(1,260)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	8.940	(1,000)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	les.	1.976	(1221)	Ordinary
0.Apr.14	I BC Contrities Australia Ltd	1100	10.505	14 1047	
Apr. 14	1 190 Commission Assessed Lead	17-5	000,0	(40)	o o
00-Apr-14	UDS SECURINES AUSITALIA	iia.	331,	(571)	Ordinary
0-Apr-14	UBS Securities Australia Ltd	Setl	16,360	(1,830)	Ordinary
0-Apr-14	UBS Securities Australia Ltd	Sell	12,516	(1,400)	Ordinary
30-Apr-14	UBS Securities Australia Ltd	Sell	26,820	(3,000)	Ordinary
0-Apr-14	UBS Securities Australia Ltd	Sell	80	Ó	Ordinary
30-Apr-14				• • • • • • • • • • • • • • • • • • • •	5
30-Apr-14	I I I I I I I I I I I I I I I I I I I	1/21	10010	(C. 41.7)	1

UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell	146,072 44,450 253,978	(16,431)	
	44,450 253,978		
	253,978		
			l
	7,014	(789)	Ordinary
	70/		
	117.045	2,000	Ordinary
	7.161		Ordinary
	35,540	4,000	Ordinary
	295		Ordinary
	2,181		Ordinary
	7,733		Ordinary
	43,806		Ordinary
	23,255)S(C)	Ordinary
UBS Securities Australia Ltd Sell	43 806	(00 47)	
	16,646		
	27	(3)	Ordinary
UBS Securities Australia Ltd Sell	134	(12)	Ordinary
	43,806	(4,900)	Ordinary
	16,977	(668'1)	Ordinary
	6	(1)	Ordinary
ies Australia Ltd Seli	4,720		
	16,932	(1,894)	
	43,806		
	3,156	(353)	Ordinary
ies Australia Ltd	16,798		Ordinary
UBS Securities Australia Ltd	13,794		Ordinary
	2,745	(307)	Ordinary
Upo Securities Australia Ltd	Soc.	(01)	Ordinary
UBS Securities Australia Ltd	286,1	(//1)	Ordinary
	47,247	(4/5)	Ordinary
USS Securities Australia I td	375	(0/)	Ordinary
	σ	(1)	Ordinary
	4.247	(475)	Ordinary
	6,249	(669)	Ordinary
	5,391	(E09)	Ordinary
UBS Securities Australia Ltd Sell	33,078	(3,700)	Ordinary
	2,691	(301)	Ordinary
	60001	(1,129)	Ordinary
lies Australia Ltd Sell	1,144	(128)	Ordinary
UBS Securities Australia Ltd Sell	2,682	(300)	Ordinary
	1,225	(137)	Ordinary
ies Australia Ltd	4,470	(200)	Ordinary
	13,410	(1,500)	Ordinary
UBS Securities Australia Ltd Self	814	(91)	Ordinary
ities Australia Ltd Sell	4,247	(475)	Ordinary
UBS Securities Australia Ltd Sell	5,373	(601)	Ordinary
ities Australia Ltd Sell	3,433	(384)	Ordinary
	10,451	(1,169)	Ordinary
	2,754	(308)	Ordinary
UBS Securities Australia Ltd Sell	295	(33)	Ordinary
	5.373	(601)	Ordinary
	8.520	(623)	Ordinary
UBS Securities Australia Ltd Sell	259	(53)	Ordinary
	15,484	(1.732)	Ordinary
	5.337	(282)	Ordinary
	1,037	(116)	Ordinary
	396	(268)	Ordinary
IRS Securities Australia 11d	1 037	(116)	Ordinary
	670	(100)	Oldinary
ODS Securities Australia Ltd	040	(100)	Ordinary
	12,123	(acc'1)	Ordinary
UBS Securities Australia Ltd	5,105	(571)	Ordinary

02-IVIdy-14	М	Seil	DOC'-	(140)	diam'r.
- 1	UBS AG London Branch	Ne l	5,321	(603)	Ordinary
02-May-14	UBS AG London Branch	E S	6 178	(91)	Ordinary
1	UBS AG London Branch	3	4 298	(787)	Ordinary
72-May-14	LIRS AG London Branch		357.0	(1010)	Ordinary Ordinary
02-Mav-14	UBS AG London Branch	100	18	000	Ordinary Ordinary
ay-14	UBS AG London Branch	lle S	168	(01)	Ordinary
02-May-14	UBS AG London Branch	Sell	22.308	(2 525)	Ordinary
32-May-14	UBS AG London Branch	≡eS	2 288	(254)	Ordinary
02-May-14	UBS AG London Branch	ll as	308	(32)	Ordinary
J2-May-14	UBS AG London Branch	9	6612	(748)	Ordinan
fay-14	UBS AG London Branch		769	(87)	Ordinary
02-May-14	UBS AG London Branch	3	513	(28)	Ordinary
fay-14	UBS AG London Branch	Sell	88	(10)	Ordinary
02-May-14	UBS AG London Branch	Sell	5.905	(899)	Ordinary
02-May-14	UBS AG London Branch	Sell	230	(26)	Ordinary
02-May-14	UBS AG London Branch	195	752	(85)	Ordinary
lay-14	UBS AG London Branch	3	4.573	(517)	Ordinary
02-May-14	UBS AG London Branch	100	13 984	(1 581)	Ordinary
02-May-14	UBS AG London Branch	100	584	(99)	Ordinary
02-May-14	UBS AG London Branch	100	2512	(284)	Ordinary
02-May-14	UBS AG London Branch	ll d'S	3 600	(407)	Ordinary
1	UBS AG London Branch		2,026	(220)	Ordinary
ļ.	UBS AG London Branch	ll d'S	557	(63)	Ordinary
ı	UBS AG London Branch	JI-S	27	9	Ordinary
	UBS AG London Branch	Sell	451	(51)	Ordinary
ł	UBS AG London Branch	Sell	1771	(20)	Ordinary
02-May-14	UBS AG London Branch	Sell	4 210	(927)	Ordinary
02-May-14	UBS AG London Branch	-	990 8	(745)	Ordinary
02-May-14	UBS AG London Branch	Sell	115	(13)	Ordinary
1ay-14	UBS AG London Branch	Seli	752	(82)	Ordinary
02-May-14	UBS AG London Branch	-Sell	212	(24)	Ordinary
02-May-14	UBS AG London Branch	Sell	1.654	(187)	Ordinary
02-May-14	UBS AG London Branch	ll-Sell	11,392	(1.288)	Ordinary
ı	UBS AG London Branch	Sell	1.070	(121)	Ordinary
1	UBS AG London Branch	Sell	831	(94)	Ordinary
02-May-14	UBS AG London Branch	Sell	2,600	(294)	Ordinary
1	UBS AG London Branch	<u></u>	1 053	(119)	Ordinary
02-May-14	UBS AG London Branch	3	827.71	(05/1/	Ordinary
02-May-14	JRS AG London Branch	3	106	(25)	Ordinary
02-May-14	URS AG London Branch	Sell	901	(12)	Ordinary
1 / - Ve	I DE AG London Broads	ilac	100	(91)	Ordinary
02-lv/dy-14	USC ACT CONSOL BEARCH	nec .	591,6	(584)	Ordinary
02-May-14	IBS AG Condon Branch	lies	0,245	(90/)	Ordinary
dy-1.4	OBS AG LOIGOT BEAILT	Dell.	4,052	(975)	Ordinary
OZ-IVIBY-14	UBS AG London Branch	Sell	17,728	(1,439)	Ordinary
-	UBS AG London Branch	Sell	5,841	(099)	Ordinary
- 1	UBS AG London Branch	Sell	1,832	(202)	Ordinary
02-May-14	UBS AG London Branch	Sell	27	(3)	Ordinary
- 1	UBS AG London Branch	Sell	828	(106)	Ordinary
02-May-14	UBS AG London Branch	Seil	1,283	(145)	Ordinary
02-May-14	UBS AG London Branch	Seil	669	(62)	Ordinary
02-May-14	UBS AG London Branch	Sell	2,123	(240)	Ordinary
02-May-14	UBS AG London Branch	Setl	88	(10)	Ordinary
lay-14	UBS AG London Branch	Sell	2,441	(276)	Ordinary
02-May-14	UBS AG tondon Branch	Sell	4,670	(528)	Ordinary
02-May-14	UBS AG London Branch	Sell	318	(36)	Ordinary
02-May-14	UBS AG London Branch	Sell	44	(2)	Ordinary
02-May-14	JBS AG London Branch	Selt	12,277	(1,388)	Ordinary
ay-14	JBS AG London Branch	Sell	4,308	(487)	Ordinary
٦	JBS AG London Branch	Self	3,883	(439)	Ordinary
٦	UBS AG London Branch	Zell Zell	3,281	(371)	Ordinary
- 1	JBS AG London Branch	Sell	12,728	(1,439)	Ordinary
- 1	UBS AG London Branch	Sell	3,059	(346)	Ordinary
02-May-14	JBS AG London Branch	Sell	1.7	(8)	Ordinary
	IDC A C Leader Bt	= 0			
	Jos AG COLIGON Branch	ž	0	(L)	Ordinary

Sell 19,132 Sell 4,470 Sell 4,72 Sell 1,782 Sell 1,783 Sell 42,823 Sell 7,603 Sell 7,603 Sell 1,184 Sell 1,184 Sell 42,829 Buy 895,401 Sell 494,338 Buy 895,401 Sell 2,0172 Buy 895,401 Sell 2,0172 Buy 89,401 Sell 2,005 Sell 3,005 Sell 3,005 Sell 1,033 Sell 1,033 Sell 1,033 Sell 2,007 Sell 3,007 Sell 4,554 Sell 4,554 Sell 4,554 Sell 3,675 Sell 4,555 Sell 4,564 <th>(2.140) (2.140) (3.90) (4.790) (3.90) (4.790) (3.90) (4.790) (4.790) (5.64) (4.790) (5.64) (7.13) (7.13) (7.13) (7.13) (1.141)</th> <th></th>	(2.140) (2.140) (3.90) (4.790) (3.90) (4.790) (3.90) (4.790) (4.790) (5.64) (4.790) (5.64) (7.13) (7.13) (7.13) (7.13) (1.141)	
φ ω - 4 ω	(500) (4,790) (1,390) (1,390) (1,390) (1,300)	
m m 1-4 m	(44) (866) (2000) (2000) (380) (4,590) (850) (850) (100,595) (100,	
8 8 1-4 8	(850) (920)	
m m m m m m m m m m m m m m m m m m m	(4790) (4790) (4790) (850) (850) (719) (719) (719) (719) (719) (719) (729) (729) (7294	
800 40	(4,790) (854) (182) (183) (713) (713) (713) (713) (713) (713) (713) (723) (724) (724) (725)	
000	(850) (850) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132) (7132)	
	(850) (132) (132) (2.099) (100,995)	
	(564) (132) (100,935) (100	
	(7139) (7139) (7139) (7139) (7130) (7130) (7130) (723) (723) (723) (723) (723) (723) (723) (723)	
	(7.132) (7.132) (1.00,995) (1.00,	
	(2.095) (100,995	
	(100,995) 100,995 19,796 (55,879) (30,095 (500)	
	(5.106) (5.106) (5.106) (5.8.73) (100.995 (5.300) (5.00) (5.00) (6.204) (1.31) (1.141)	
	(5,106) (19,796 (3,879) (3,879) (3,800) (5,000) (5,000) (5,000) (5,000) (3,241) (119) (119) (119) (119) (1185)	
	(55,879) (65,879) (63,309) (500) (50	
	(55,879) (100,995) (100,995) (500) (500) (500) (500) (500) (1019) (119)	
	(3,927) (5,000) (5,000) (5,000) (5,000) (5,000) (6,000) (1,000	
	(3,300) (9,294 (500) (500) (500) (2,264) (119) (119) (119) (118) (18) (18) (18) (18)	
	(500) (500) (500) (500) (500) (500) (7.264) (7.264) (7.264) (7.85) (7.85) (7.85)	
	(500) (500) (500) (2,264) (1,264) (1,19) (1,19) (1,19) (1,18) (1,18) (1,18) (1,18) (1,18) (1,18) (1,18) (1,18) (1,18) (1,18)	
	(500) 956 (2,264) (1,264) (1,19) (1,19) (1,18) (2,29) (7,85) (3,27)	
20,059 28,683 28,683 1,053 3,080 2,037 6,932 34,675 89 4,564 4,564	956 (2,264) (3,241) (119) (119) (239) (3,927) (3,927)	
20.059 20.059 1,053 3,080 3,080 2,027 6,932 34,675 89 4,564 4,564	(2,264) (3,241) (119) (119) (348) 229 (785) (3,927)	
28.683 1.053 3.080 2.027 6.932 3.4675 4.567 2.875	(3,241) (119) (348) (229 (785) (3,927)	
1,033 3,080 2,027 6,332 34,675 89 4,564 4,564 2,875	(119) (348) 229 (785) (3,927)	
3,080 2,027 6,932 34,675 89 4,564 2,875	(348) 229 (785) (3,927)	
2,027 6,932 34,675 89 89 4,564 2,875	(785) (3,927)	
6,932 34,675 89 4,564 2,875	(3,927)	
34,675 89 8,564 2,875	(3,927)	Ordinary
89 4,564 2,875		
4,554	01	Ordinary
2,0/3	(916)	Ordinary
2436	(676)	(Jeuga)
0,470	(757)	Ordinary
1 831	(202)	Ordinaly Ordinaly
53	(707)	Ordinary
503	(0)	Ordinary
000	(/0)	Ordinary
677'1	(881)	Ordinary
3,761	(425)	Ordinary
3,319	(375)	Ordinary
5,142	(581)	Ordinary
2,726	(308)	Ordinary
7,218	(816)	Ordinary
6,209	(702)	Ordinary
849	(96)	Ordinary
611	(69)	Ordinary
116,1	(216)	Ordinary
4,803	(543)	Ordinary
761	(98)	Ordinary
4 219	(477)	Ordinary
000 8	(470)	Countries
1,000	(Cro)	Cloring
/81'/	(813)	Orginary
2,387	(270)	Ordinary
2,937	(332)	Ordinary
1,309	(148)	Ordinary
62	(6)	Ordinary
2,859	(068)	Ordinary
168	(19)	Ordinary
009	(89)	Ordinary
274	(31)	Ordinary
	(31)	Ordinan
	(0.)	O'CILINGIA CICILINGIA
	5,142 7,218 7,218 849 6,209 849 6,191 1,911 1,911 4,219 4,060 4,060 4,060 7,187 2,337 1,309 1,309 7,819 7,819 1,309 7,819 7,819 1,309 7,819 1,309 7,819 1,309	

U2-May-14 UB 02-May-14 UB 02-Ma	UBS AG London Branch UBS AG London Branch	Self	77.	(88)	Ordinary
	55 Ad London branch			(43)	
1 1 1 1 1 1 1 1	Se Annual Design	- R	380	(AL)	Ordinary
	BS Ale London Branch	les	318	(36)	Ordinary
	8S AG London Branch	Sell	592	(67)	Ordinary
	UBS AG London Branch	Sell	159	(18)	Ordinary
1 1	3S AG London Branch	Seil	2.603	(295)	Ordinary
1 1 1 1	UBS AG London Branch	Sell	6.178	(002)	Ordinary
1 1 1	UBS AG London Branch	Sell	2.505	(784)	Ordinary
1 1	3S AG London Branch	Jes.	2 648	(305)	Ordinary
1	UBS AG London Branch	las	2 648	(300)	Ordinan
	35 AG London Branch	19	1,050	(102)	ording.
ł	MA AG London Bronch	100	000,1	(10/)	Cidillary
T	100 AC London Description		061,1	(871)	Ordinary
1	33 Ad Editori Braildi	<u> </u>	53	(g)	Ordinary
- 1	UBS Ats London Branch	II.	988	(38)	Ordinary
02-May-14 UB	35 AG London Branch	Sell	1,050	(119)	Ordinary
	35 AG London Branch	Sell	4,201	(476)	Ordinary
02-May-14 UB	UBS AG London Branch	Sell	2,074	(532)	Ordinary
02-May-14 UB	UBS AG London Branch	Sell	18,930	(2,145)	Ordinary
32-May-14 UB	UBS AG London Branch	Sell	539	(61)	Ordinary
ı	UBS AG London Branch	Sell	71	(8)	Ordinary
1	35 AG London Branch	Sell	891	(101)	Ordinary
ŀ	3S AG London Branch	Call	35		Ordinary
1	IRS AG London Branch	in a	CCF	(1)	Cidinaly
T	C A C Landan Dranet	120	132	(CI)	Ordinary
T	Des Ad Edition Branch	Sell	7.26	(67)	Ordinary
Ť	So Ald London Branch	las	35	(4)	Ordinary
Ť	UBS AG London Branch	Sell	686	(112)	Ordinary
7	UBS AG London Branch	Sell	53	(9)	Ordinary
┪	35 AG London Branch	Sell	606	(103)	Ordinary
02-May-14 UB	UBS AG London Branch	Sell	238	(61)	Ordinary
F	JBS AG London Branch	Seil	888	(01)	Ordinary
Ī	UBS AG London Branch	Į o	2 480	(282)	Codingo
T	I BS A G London Branch		1 104	(10.7)	O Colonial y
1	AC AG London Branch	100	4000	(50)	Clumary
	SCACLES CONTROL	100	677	(07)	Orginary
02-May-14 UB	UBS Au London Branch	Sell	6	(E)	Ordinary
- 1	35 AG London Branch	Sell	1,041	(118)	Ordinary
7	JBS AG London Branch	Sell	006	(102)	Ordinary
- 1	UBS AG London Branch	Seli	5,251	(265)	Ordinary
	35 AG London Branch	Sell	635	(72)	Ordinary
	3S AG London Branch	Sell	185	(21)	Ordinary
02-May-14 UB	UBS AG London Branch	Se.	σ	(2)	Ordinary
	IRS AG London Branch	Coll	7 393	(1)	Ording)
T	IDC AG London Dranch	100	100	(1)	O G
Ť	SO AG EGILGOI BIAILOI	nec.	468	(54)	Ordinary
7	UBS AG London Branch	Sell	459	(52)	Ordinary
02-May-14 UB	UBS AG London Branch	Seli	35	(4)	Ordinary
	35 AG London Branch	Sell	318	(36)	Ordinary
Г	UBS AG London Branch	Sell	733	(83)	Ordinary
T	UBS AG London Branch	ll d'S	848	(96)	Ordinan
Ť	IRS AG London Branch	le le	100	(10)	Cicilian y
T	100 A C London Drawsh	100	CBI	(17)	Clumary
T	S AC ECITOR DISTRICT	IIAC II	n ((1)	Ordinary
00 11 14 00	des Ad Loridor Blanch	llac l	n .	Ê	Ordinary
1	S AG FORGON Branch	Jes.	88	(010)	Ordinary
7	JBS AG London Branch	Seil	1,925	(218)	Ordinary
02-May-14 UB:	UBS AG London Branch	Sell	194	(22)	Ordinary
32-May-14 UB:	UBS AG London Branch	Sell	450	(51)	Ordinary
Ť	S AG London Branch	Spl	8 154	(PCD)	Ordinary
Ť	JBS AG London Branch	lg.	247	(80)	Ordinan
t	BS AG London Branch	Ilo	2000	(02)	or diagram
T	S AC ECHOOL SIGNAL	361	210,2	(077)	Orginary
T	DES AG LORIGON Branch	Sell Sell	1,033	(/11)	Ordinary
1	UBS AG London Branch	Sell	37/	(37)	Ordinary
٦	UBS AG London Branch	Sell	221	(22)	Ordinary
_	S AG London Branch	Sell	230	(9)	Ordinary
02-May-14 UB:	JBS AG London Branch	les	265	(30)	Ordinary
32-May-14 UB	JBS AG London Branch	Seil	468	(23)	Ordinary
۳	UBS AG London Branch	le S	432	(66)	Ordinary
Т	S A Carden Branch	130	100	(10)	oldingi,
UZ-IMAy-14 UB:	UBS AG London Branch	llec	177	(52)	Ordinary

02-May-14	I IBS AG London Branch	Call	515	90	L
2-May-14	UBS AG London Branch	<u> </u>	610	(54)	
02-May-14	UBS AG London Branch	Set	822	(55)	Ordinary
2-May-14	UBS AG London Branch	Sell	265	(29)	Ordinary
02-Mav-14	UBS AG London Branch	الح	150	(31)	
02-Mav-14	PRS AG London Branch	3	6	(0)	Ordinary
02-May 14	I I O A C I condon Dranch	170	6		
02-May 14	Ind AG London Branch	los	000	000	
May 14	IBC AG London Branch	li con	010		
	de la companion de la companio	125	21.0		
02-May-14	UBS AG LONGON BRAINCH	Sell	5/4	(31)	
-IVIdy-14	UBS AG Loridon Branch	Sell	/7	(3)	╛
02-May-14	UBS AG London Branch	Sell	530	(09)	Ordinary
02-May-14	UBS AG London Branch	Sell	80	(6)	Ordinary
02-May-14	UBS AG London Branch	Sell	17	(8)	Ordinary
-May-14	UBS AG London Branch	Sell	964	(601)	Ordinary
02-May-14	UBS AG London Branch	Sell	9	(2)	Ordinary
02-Mav-14	UBS AG London Branch	Sell	44	(5)	Ordinary
	UBS AG London Branch	Self	44	(2)	Ordinary
	LIBS AG London Branch	Soll	150	(17)	Ordinan
12-May-14	I IBC AG London Branch	Coff	000	(44)	
ı	IBO AG London Branch	100	2002		
(vidy-15		1341	7527		Cremary
JZ-May-14	UBS AG London Branch	Sell	756	(53)	Ordinary
2-May-14	UBS AG London Branch	Sell	9		Ordinary
-May-14	UBS AG London Branch	Sell	336	(38)	Ordinary
-May-14	UBS AG London Branch	Sell	645	(23)	Ordinary
2-May-14	UBS AG London Branch	Sell	1 494	(169)	Ordinan
-Mav-14	IBS AG London Branch	lley	σ	(1)	Ordinan
12 May 14	I DC ACT ondon Brooch	Coll	000	(10)	
INION-14	מפס אם בתו אמון פושוקו		057	(20)	
02-May-14	UBS AG London Branch	Self	601	(68)	
02-May-14	UBS AG London Branch	Sell	71	(8)	Ordinary
-May-14	UBS AG London Branch	Sell	239	(61)	Ordinary
-May-14	UBS AG London Branch	Sell	8,830	(1,000)	Ordinary
-May-14	UBS AG London Branch	Sell	2.199	(576)	Ordinary
2-May-14	UBS AG London Branch	Sel	299	(75)	Ordinary
8420114	I IBC A C London Broach	153	2000	(110)	de la la
z ividy i +	COS AC LONGOI DIRICII	11-0	767'7	(00)	Cidilary
Z-IVIdy- 14		IIAC	717	(74)	Ordinary
-May-14	UBS AG London Branch	Sell	79	(6)	Ordinary
2-May-14	UBS AG London Branch	Sell	18	(2)	Ordinary
J2-May-14	UBS AG London Branch	Sell	1881	(213)	Ordinary
12-May-14	LIBS AG London Branch	الع	7.060	(800)	Ordinary
1420.14	100 AC London Dranch	100	200,1	(0)	, in the second
z-ivlay-14	UBS AG London Branch	Seil	97	(5)	Ordinary
-May-14	UBS AG London Branch	Sell	5,269	(597)	Ordinary
)2-May-14	UBS AG London Branch	Sell	8,825	(1,000)	Ordinary
02-May-14	UBS AG London Branch	3	2 055	(533)	Ordinary
NASSE 14	UBC ACT condon Branch	100	500	(50)	Coding
U2-IVIdy-14	UBS AG London Branch	Sell	507	(23)	Ordinary
02-May-14	UBS AG London Branch	Sell	759	(86)	Ordinary
	UBS AG London Branch	Sell	1,986	(225)	Ordinary
1	UBS AG London Branch	Seil	1 491	(169)	Ordinary
14 April 14	O A Condens Describe	1100	04.7	(001)	(min)
П	UDS AG LUINGII BIAIICI)eii	9/1'0	(700)	Ordinary
١	UBS AG London Branch	Seli	1,986	(225)	Ordinary
02-May-14	UBS AG London Branch	Sell	8,825	(1,000)	Ordinary
02-May-14	UBS AG London Branch	9	1 774	(100)	Ordinary
	DC AC Landan Daniel	100	000	(104)	Comman of
- inlay	Opp and collection	II ac	007	(/7)	Clariary
)2-May-14	UBS AG London Branch	Sell	274	(31)	Ordinary
32-May-14	UBS AG London Branch	Sell	086	(111)	Ordinary
02-May-14	LiRS AG London Branch	[]	1,4	(8)	Ordinan
1 - May - 1 4	FIG. ACT and a Branch	100	010	(011)	Codings
ividy-14	ממו של המו המו המו המו המו המו המו המו המו המו	ii.	000'1	(611)	Cicilary
12-May-14	UBS AG London Branch	Set	5/4	(65)	Ordinary
-May-14	UBS AG London Branch	Sell	353	(40)	Ordinary
02-May-14	UBS AG London Branch	Sell	768	(87)	Ordinary
02-Mav-14	UBS AG London Branch	Sell	442	(50)	Ordinary
		= 0		(95)	Countries
UZ-May-14	UBS AG London Branch	Sell	6	(1)	Ordinary
-May-14	UBS AG London Branch	Sell	309	(32)	Ordinary
-May-14	UBS AG London Branch	Sell	477	(54)	Ordinary
12-May-14	LIRS AG London Branch	le S	1/23	(65)	Ordinan
Т			100	(co)	3 1 1 1 1
	UBS AG London Branch	Sell	380	(43)	Ordinary
Г	I IRS &G London Branch	Cell	29	(2)	Ordinary
7	מפו את המושות היו	lacil.	30	1/./	Cicinaly

05-May-14	I IRS AG London Branch	log	-	()	
av-14	UBS AG London Branch	13/3	2 038	(228)	A COLOR
05-May-14	UBS AG London Branch	Sell	331	(37)	Ordinary
05-May-14	UBS AG London Branch	Sell	768	(1001)	Ordinary
05-May-14	UBS AG London Branch	9	2 100	(246)	Ordinary
05-Mav-14	UBS AG London Branch	9	3.853	(121)	Ordinary
1ay-14	UBS AG London Branch	Sell	4.917	(550)	Ordinary
5-May-14	UBS AG London Branch	Sell	5.060	(566)	Ordinary
5-May-14	UBS AG London Branch	Sell	1 350	(151)	Ordinary
35-May-14	UBS AG London Branch	Sell	2,880	(322)	Ordinary
05-May-14	UBS AG London Branch	Sell	814	(16)	Ordinary
35-May-14	UBS AG London Branch	Setl	1,226	(137)	Ordinary
05-May-14	UBS AG London Branch	Sell	5,468	(611)	Ordinary
Aay-14	UBS AG London Branch	Sell	493	(55)	Ordinary
05-May-14	UBS AG London Branch	Sell	1,729	(193)	Ordinary
05-May-14	UBS AG London Branch	Sell	5,645	(089)	Ordinary
5-May-14	UBS AG London Branch	Sell	4,408	(492)	Ordinary
5-May-14	UBS AG London Branch	Sell	4,935	(552)	Ordinary
√ay-14	UBS AG London Branch	Sell	3,419	(382)	Ordinary
05-May-14	UBS AG London Branch	Seli	1,459	(163)	Ordinary
05-May-14	UBS AG London Branch	Sell	4,801	(537)	Ordinary
05-May-14	UBS AG London Branch	Sell	4,801	(537)	Ordinary
05-May-14	UBS AG London Branch	Sell	4,260	(477)	Ordinary
5-May-14	UBS AG London Branch	Sell	643	(72)	Ordinary
5-May-14	UBS AG London Branch	Sell	2,090	(234)	Ordinary
5-May-14	UBS AG London Branch	Sell	666	(112)	Ordinary
05-May-14	UBS AG London Branch	Sell	5,316	(296)	Ordinary
05-May-14	UBS AG London Branch	Sell	2,375	(392)	Ordinary
05-May-14	UBS AG London Branch	Sell	4,947	(554)	Ordinary
5-May-14	UBS AG London Branch	Sell	2,188	(242)	Ordinary
May-14	UBS AG London Branch	Sell	5,135	(575)	Ordinary
05-May-14	UBS AG London Branch	Sell	2,640	(562)	Ordinary
day-14	UBS AG London Branch	Sell	3,764	(422)	Ordinary
35-May-14	UBS AG London Branch	Sell	2,027	(227)	Ordinary
35-May-14	UBS AG London Branch	Sell	269	(28)	Ordinary
5-May-14	UBS AG London Branch	Sell	3,688	(413)	Ordinary
5-May-14	UBS AG London Branch	Sell	3,397	(380)	Ordinary
5-May-14	UBS AG London Branch	Seli	2,280	(255)	Ordinary
Aay-14	UBS AG London Branch	Sell	3,406	(381)	Ordinary
5-May-14	UBS AG London Branch	Sell	5,105	(571)	Ordinary
05-May-14	UBS AG London Branch	Sell	3,020	(337)	Ordinary
05-May-14	UBS AG London Branch	Sell	949	(106)	Ordinary
5-May-14	UBS AG London Branch	Sell	1,307	(146)	Ordinary
Aay-14	UBS AG London Branch	Sell	2,390	(267)	Ordinary
05-May-14	UBS AG London Branch	Seil	2,130	(238)	Ordinary
05-May-14	UBS AG London Branch	Sell	1,665	(186)	Ordinary
05-May-14	UBS AG London Branch	Sell	3,468	(387)	Ordinary
05-May-14	UBS AG London Branch	Sell	923	(103)	Ordinary
5-May-14	UBS AG London Branch	Sell	851	(95)	Ordinary
5-May-14	UBS AG London Branch	Sell	27	(3)	Ordinary
/ay-14	UBS AG London Branch	Sell	735	(82)	Ordinary
05-May-14	UBS AG London Branch	Self	1,855	(202)	Ordinary
05-May-14	UBS AG London Branch	Sell	2,930	(327)	Ordinary
35-May-14	UBS AG London Branch	Sell	4,740	(529)	Ordinary
Aay-14	UBS AG London Branch	les	5,304	(265)	Ordinary
05-May-14	UBS AG London Branch	Sell	1,676	(187)	Ordinary
05-May-14	UBS AG London Branch	Sell	177	(98)	Ordinary
Aay-14	UBS AG London Branch	Sell	698	(26)	Ordinary
5-May-14	UBS AG London Branch	Sell	4,794	(535)	Ordinary
/ay-14	UBS AG London Branch	Sell	5,477	(612)	Ordinary
5-May-14	UBS AG London Branch	Sell	2,724	(304)	Ordinary
05-May-14	UBS AG London Branch	Sell	2,132	(238)	Ordinary
35-May-14	UBS AG London Branch	Sell	91/	(08)	Ordinary
05-May-14	UBS AG London Branch	Sell	061'9	(269)	Ordinary
				(200)	, ,,,,,,
U5-May-14	JUBS AG London Branch	Sell	54	(9)	Ordinary

UBS AG London Branch				Ordinan
	Sell	150	(/1)	Ordinary
UBS AG London Branch	Sell	1,492	(169)	Ordinary
UBS AG London Branch	Sell	3,046	(345)	Ordinary
S AG London Branch	Sell	265	(30)	Ordinary
UBS AG London Branch	Sell	3,285	(372)	Ordinary
JBS AG London Branch	Sell	494	(26)	Ordinary
UBS AG London Branch	Sell	4,148	(470)	Ordinary
35 AG London Branch	Seif	8,551	(696)	Ordinary
UBS AG London Branch	Sell	29.634	(3.358)	Ordinary
UBS AG London Branch	Sel	3.558	(403)	Ordinary
IRS AG London Branch	3	2026	(64)	i i i i
DE ACTORDO DESCRIP	llac	300	(40)	Ordinary
DC ACT and a branch	100	1,0	(66)	Clumary
SS AG London Branch	Sell Sell	212	(85)	Ordinary
JBS AG London Branch	Sell	212	(24)	Ordinary
35 AG London Branch	Sell	62	(2)	Ordinary
UBS AG London Branch	Sell	6	(1)	Ordinary
3S AG London Branch	Sell	62	(6)	Ordinary
UBS AG London Branch	Sell	141	(16)	Ordinary
35 AG London Branch	llas	327	(7.57)	Ordinan
IBS AG London Branch			(1)	Ordinal y
DO ACIDITION DISINI	1120	200	(/)	Ordinary
UBS AG London Branch	Sell	506	(103)	Ordinary
JBS AG London Branch	Sell	380	(43)	Ordinary
UBS AG London Branch	Sell	256	(105)	Ordinary
35 AG London Branch	Sell	1.448	(164)	Ordinary
35 AG London Branch	i d'y	153	(50)	Ordinary
I IBS AG London Branch	03	787	(22)	Ordinan
IBC ACT corden Property	Soli		(60)	Ordinary Ordinary
SA AG EGINGOL BLANCK	liaci	n (,,	(1)	Ordinary
OBS AG LONGOR BRANCH	Seli	910,1	(183)	Ordinary
JBS AG London Branch	Sell	1,298	(147)	Ordinary
	Sell	138,326	(15,701)	Ordinary
JBS AG London Branch	Sell	176,746	(20,062)	Ordinary
35 AG London Branch	Sell	249,393	(28,308)	Ordinary
UBS AG London Branch	Stock returned	¥	(4,293)	Ordinary
UBS AG London Branch	Stock returned	ANN	(1.040)	Ordinary
JBS AG London Branch	Stock returned	Δ/N	(18 935)	Ordinary
IRC AG London Branch	Stock returned	S Z	(000 0)	Ordinan
DO Contraction Australia (4)		0,000	445	O'calina)
Obs securities Australia Ltu	ing	3,342	445	Orginary
85 Securities Australia Ltd	Sell	23,452	(2,638)	Ordinary
JBS Securities Australia Ltd	Sell	137,600	(15,535)	Ordinary
JBS Securities Australia Ltd	Sell	48,299	(5,447)	Ordinary
JBS Securities Australia Ltd	Bov	79.600	000.6	Ordinary
35 Securities Australia I to	(Sel	5110	(580)	Ordinary
DC Countries Australia Led	100	0 03	(1000)	Ordinary Ordinary
23 Seculities Australia Liu	ilac i	660,0	(215)	Gerrary
JBS Securifies Australia Ltd	Seli	555	(63)	Ordinary
UBS Securities Australia Ltd	Sell	2,185	(247)	Ordinary
JBS Securities Australia Ltd	Sell	4,420	(200)	Ordinary
UBS Securities Australia Ltd	Sell	4,420	(200)	Ordinary
UBS Securities Australia Ltd	Sell	2.413	(1577)	Ordinary
RS Securities Australia Ltd	9	1 020	(101)	Ordinan
IDC Commission Assessed to and	===	0.00	(12.1)	t in it
po seculties wasti alla Ltd	i oci	5,210	(304)	Ordinary
UBS Securities Australia Ltd	Seil	17,450	(1,9/4)	Ordinary
85 Securities Australia Ltd	Sell	186	(21)	Ordinary
UBS Securities Australia Ltd	Buy	20,623	2,325	Ordinary
JBS Securities Australia Ltd	Buy	2.696	304	Ordinary
IRS Sequitible Australia Ltd	Brov	o	-	Ordinary
IRS AG London Branch	19	654	(27)	Ordinan
In A C London Dunch	=======================================	0,100	(10)	O'dirini
S AG LOTIGOTI BILATICI	- X	9,102	(190)	Ordinary
JBS AG London Branch	lle.	6,/56	(/54)	Ordinary
JBS AG London Branch	Sell	5,802	(649)	Ordinary
JBS AG London Branch	Sell	2,802	(649)	Ordinary
UBS AG London Branch	Sell	6	(1)	Ordinary
S AG London Branch	Sel	62	6	Ordinary
S AG London Branch	90	818	(60)	Ordinan
COS ACIDION DISTRICT	900	0 0 0	(60)	Cidiliary
35 AG London Branch	Sel	180,4	(595)	Ordinary
UBS AG London Branch	Sell	643	17.6	
		1	(7/)	Ordinary

referint Derivative (NA 2500) y inferint Derivative (NA 2500) y inferint Derivative (NA 2500) y inferint (112) y inferint (112) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (113) y inferint (12,629) y inferint (12,639) y inferint (12,639) y inferint (12,639) y inferint (12,639) y inferint (13,639) y inferint (13,639) y inferint (13,639) y inferint (116) y inferint (11			Enter Into Derivative	NA	2,500	Optio
Bits Weight Management Australia Ltd David Management Australia Ltd Inventor Australia Ltd A 200 165 Sweith Management Australia Ltd Self 1,192 1,132 165 Securies Australia Ltd Self 1,102 1,132 165 Securies Australia Ltd Buy 2,542 284 165 Securies Australia Ltd Buy 2,542 2,6410 165 Securies Australia Ltd Buy 2,6420 2,6410 165 Securies Australia Ltd Buy 2,6420 2,6410 165 Securies Australia Ltd Buy 2,6420 2,6410 165 Securies Australia Ltd Buy 2,626 2,6410 165 Securies Australia Ltd Buy 2,626 2,6410 165 Securies Australia Ltd Buy 2,626 2,626 165 Secur			200	A 200 C	351	S C
1985 Securities Australia Ltd Self		ustralia Ltd ustralia Ltd	Ruy	415/1		
1982 1982		ustralia Ltd	Transfer-In	AVA	4 000	Sign
Securites Australia Ltd Self 1,000 1,0			الع	1 192	(133)	
Big Securities Australia Ltd Self Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd Self Securities Australia Ltd			9	1 002	(212)	
Big Securities Australia LH Self 1979 12,677 12,629 10,53 Securities Australia LH Self 1979 12,6271 12,629 10,53 Securities Australia LH Self 10,579 12,6271 12,030 10,53 Securities Australia LH Self 10,5271 12,030 1			Sirv	200,7	700	
USS Securities Australia Ltd Self 123,477 12,629 10,006			Birv	909	67	
Bits Securities Australia Ltd Sell 2.5.5.15 3.07.81 Lobis Securities Australia Ltd Buly 2.5.5.15 3.07.81 Lobis Securities Australia Ltd Buly 2.5.4.20 1.0.2030 Lobis Securities Australia Ltd Buly 3.6.4.20 2.6.4.10 Lobis Securities Australia Ltd Buly 7.1.20 8.0.00 Lobis Securities Australia Ltd Buly 4.0.26 2.6.4.10 Loss Walth Management Australia Ltd Buly 4.0.26 2.6.4.10 Loss Walth Management Australia Ltd Sell 3.6.5.22 (4.0.559) Loss Walth Management Australia Ltd Sell 3.0.0.6 (5.7.1) Loss Walth Management Australia Ltd Sell 3.0.0.6 (5.7.1) Loss Securities Australia Ltd Sell 3.0.0.6 (5.7.1) Loss Securities Australia Ltd Sell 3.0.0.6 (5.7.1) Loss Securities Australia Ltd Sell 3.0.0.6 (5.7.1) Loss Securities Australia Ltd Sell 3.0.0.6 (5.7.1) Loss Securities Australia Ltd Sell <td></td> <td></td> <td>Coll</td> <td>777 66</td> <td>1000</td> <td></td>			Coll	777 66	1000	
Bis Securities Australia LIM Self 1,03,08 7,030 1085 Securities Australia LIM Self 1,03,092 7,030 1085 Securities Australia LIM Self 286,430 16,510 1085 Securities Australia LIM Self 363,092 16,510 1085 Securities Australia LIM Buy 1,102 18,000 1085 Securities Australia LIM Self 1,002 18,000 1085 Weathin Management Australia LIM Self 1,002 1,002 1085 Weathin Management Australia LIM Self 1,002 1,002 1085 Weathin Management Australia LIM Self 1,002 1,002 1085 Weathin Management Australia LIM Self 66,447 1,002 1085 Securities Australia LIM Self 1,002 1,116 1085 Securities Australia LIM Self 1,002 1,113 1085 Securities Australia LIM Self 1,002 1,143 1085 Securities Australia LIM Self 1,002 1,143 1085 Securities Australia LIM Self 1,002 1,143			وا	27.55	(27,042)	
Obs. Securities Australia Ltd Self 10,2929 (17,202) LOSS Securities Australia Ltd Bluy 356,420 (16,039) LOSS Securities Australia Ltd Bluy 71,220 (16,039) LOSS Securities Australia Ltd Bluy 71,220 (16,039) LOSS Securities Australia Ltd Bluy 71,220 (16,039) LOSS Securities Australia Ltd Bluy 4,996 5521 LOSS Wealth Narragement Australia Ltd Self 6,228 (17,17) LOSS Wealth Narragement Australia Ltd Self 6,228 (7,417) LOSS Wealth Narragement Australia Ltd Self 6,228 (7,417) LOSS Securities Australia Ltd Self 1,000 (1,134) LOSS Securities Australia Ltd Self 1,26,78 (1,69) LOSS Securities Australia Ltd Self 1,26,78 (1,69) LOSS Securities Australia Ltd Self 1,26,78 (1,69) LOSS Securities Australia Ltd Self 2,332 (1,69) LOS Securities Australia Ltd Self 2,345			- A	027.03	70,007	
USS Securities Australia Ltd Sell Sell Seld			oug.	00/,20	050,7	Orgina
Obs. Securities Australia Ltd Sell 35,002 (4,0,593) UBS Securities Australia Ltd Bloy 71,220 (4,0,593) UBS Securities Australia Ltd Bloy 1,1,220 (8,000 UBS Securities Australia Ltd Bloy 4,956 (524) UBS Wealth Management Australia Ltd Sell 6,328 (7,11) UBS Securities Australia Ltd Sell 6,328 (7,11) UBS Securities Australia Ltd Sell 1,060 (11,343) UBS Securities Australia Ltd Sell 8,079 (1,091) UBS Securities Australia Ltd Sell 1,090 (1,108) UBS Securities Australia Ltd Sell 2,378 (1,090) UBS Securities Australia Ltd Sell 3,283 (3,004) UBS Securities Australia Ltd Sell 3,283 (3,004)			D. III	9/8/01	(050,21)	Gana
USS Securities Australia Ltd Sell 73,202 (40,539) UBS Securities Australia Ltd Buy 71,202 (40,539) UBS Securities Australia Ltd Buy 71,202 (524) UBS Weachth Management Australia Ltd Buy 4,539 (524) UBS Wealth Management Australia Ltd Sell 7,622 (116) UBS Wealth Management Australia Ltd Sell 7,042 (116) UBS Securities Australia Ltd Sell 5,043 (116) UBS Securities Australia Ltd Sell 5,046 (5,73) UBS Securities Australia Ltd Sell 1,042 (11,43) UBS Securities Australia Ltd Sell 5,056 (5,73) UBS Securities Australia Ltd Sell 1,28,78 (14,73) UBS Securities Australia Ltd Sell 2,396 (1,13) UBS Securities Australia Ltd Sell 3,289 (6,24) UBS Securities Australia Ltd Sell 3,289 (1,13) UBS Securities Australia Ltd Sell 3,289 (1,13)			Buy	736,420	26,410	Ordina
UBS Securities Australia Ltd BMy 17,220 8,00 UBS Securities Australia Ltd BMy 141,621 15,825 UBS Wealth Management Australia Ltd Sell 4,996 551 UBS Wealth Management Australia Ltd Sell 6,623 88.1 UBS Wealth Management Australia Ltd Sell 1,622 88.1 UBS Wealth Management Australia Ltd Sell 1,628 88.1 UBS Wealth Management Australia Ltd Sell 1,628 88.1 UBS Securities Australia Ltd Sell 1,628 6,573 UBS Securities Australia Ltd Sell 1,188.0 (1,1441) UBS Securities Australia Ltd Sell 1,28.2 (1,471) UBS Securities Australia Ltd Sell 1,28.9 (1,1441) UBS Securities Australia Ltd Sell 1,28.9 (1,1441) UBS Securities Australia Ltd Sell 2,38.3 (3,66) UBS Securities Australia Ltd Sell 3,28.3 (3,66) UBS Securities Australia Ltd Sell 3,28.3 (3,66) <td></td> <td></td> <td>Sell</td> <td>363,092</td> <td>(40,593)</td> <td>Ordina</td>			Sell	363,092	(40,593)	Ordina
UBS Weachth Australia Ltd Sell 4,980 (7,411) UBS Wealth Management Australia Ltd Sell 7,880 (521 UBS Wealth Management Australia Ltd Sell 7,880 (7,411) UBS Wealth Management Australia Ltd Sell 7,642 (7,411) UBS Wealth Management Australia Ltd Sell 5,623 (7,411) UBS Wealth Management Australia Ltd Sell 5,623 (7,411) UBS Wealth Management Australia Ltd Sell 5,623 (7,411) UBS Securities Australia Ltd Sell 1,042 (7,411) UBS Securities Australia Ltd Sell 1,890 (7,411) UBS Securities Australia Ltd Sell 1,890 (7,411) UBS Securities Australia Ltd Sell 1,890 (7,141) UBS Securities Australia Ltd Sell 1,706 (1,143) UBS Securities Australia Ltd Sell 1,776 (1,99) UBS Securities Australia Ltd Sell 1,776 (1,99) UBS Securities Australia Ltd Sell 1,776 (1,99) </td <td></td> <td></td> <td>Buy</td> <td>71,220</td> <td>8,000</td> <td>Ordina</td>			Buy	71,220	8,000	Ordina
UBS Weacht Management Australia Ltd Sell 4,896 551 UBS Wealth Management Australia Ltd Sell 4,896 (524) UBS Wealth Management Australia Ltd Sell 1,022 (116) UBS Wealth Management Australia Ltd Sell 1,024 (116) UBS Securities Australia Ltd Sell 5,046 (5,73) UBS Securities Australia Ltd Sell 10,046 (5,73) UBS Securities Australia Ltd Sell 1,043 (11,143) UBS Securities Australia Ltd Sell 1,287 (11,431) UBS Securities Australia Ltd Sell 1,287 (13,431) UBS Securities Australia Ltd Sell 1,287 (14,431) UBS Securities Australia Ltd Sell 1,287 (14,431) UBS Securities Australia Ltd Sell 2,395 (16,73) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (4,621)			Buy	141,621	15,825	Ordina
UBS Securities Australia Ltd Sell 4,680 (524) UBS Wealth Management Australia Ltd Sell 7,692 (7,411) UBS Securities Australia Ltd Sell 5,092 (7,411) UBS Securities Australia Ltd Sell 5,092 (7,411) UBS Securities Australia Ltd Sell 5,092 (7,411) UBS Securities Australia Ltd Sell 10,086 (1,143) UBS Securities Australia Ltd Sell 1,086 (1,143) UBS Securities Australia Ltd Sell 1,006 (1,143) UBS Securities Australia Ltd Sell 1,006 (1,144) UBS Securities Australia Ltd Sell 1,006 (1,144) UBS Securities Australia Ltd Sell 2,395 (1,657) UBS Securities Australia Ltd Sell 2,395 (1,657) UBS Securities Australia Ltd Sell 2,395 (1,670) UBS Securities Australia Ltd Sell 2,322 (1,670) UBS Securities Australia Ltd Sell 2,323 (1,670)		ustralia Ltd	Buy	4,936	551	Ordina
UBS Wealth Management Australa Ltd Buy 7.893 881 UBS Wealth Management Australa Ltd Sell 66,328 (7.11) UBS Securities Australia Ltd Sell 5.0046 (5.73) UBS Securities Australia Ltd Sell 5.0046 (5.73) UBS Securities Australia Ltd Sell 1.042 (1.11) UBS Securities Australia Ltd Sell 1.02,004 (5.73) UBS Securities Australia Ltd Sell 1.02,00 (3.790) UBS Securities Australia Ltd Sell 1.02,00 (3.713) UBS Securities Australia Ltd Sell 1.02 (7.02) UBS Securities Australia Ltd Sell 2.005 (3.04) UBS Securities Australia Ltd Sell 2.005 (3.72) UBS Securities Australia Ltd Sell 2.006 (1.124) UBS Securities Australia Ltd Sell 2.007 (3.02) UBS Securities Australia Ltd Sell 2.006 (1.02) UBS Securities Australia Ltd Sell 2.007 (2.04)		ustralia Ltd	Sell	4,680	(524)	Ordina
UBS Securities Australia Ltd Sell 1,042 (741) UBS Securities Australia Ltd Buy 66,328 (7,41) UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Sell 10,485 (1,573) UBS Securities Australia Ltd Sell 10,1860 (1,1,43) UBS Securities Australia Ltd Sell 10,1860 (1,1,43) UBS Securities Australia Ltd Sell 1,830 (2,04) UBS Securities Australia Ltd Sell 1,830 (2,04) UBS Securities Australia Ltd Sell 1,180 (1,05) UBS Securities Australia Ltd Sell 1,175 (1,05) UBS Securities Australia Ltd Sell 1,175 (1,05) UBS Securitie		ustralia Ltd	Buy	7,893	188	Ordina
UBS Securities Australia Ltd Sell 1,042 (116) UBS Securities Australia Ltd Sell 5,046 (5,53) UBS Securities Australia Ltd Sell 34,034 (3,790) UBS Securities Australia Ltd Sell 10,1860 (11,1343) UBS Securities Australia Ltd Sell 10,1860 (11,1343) UBS Securities Australia Ltd Sell 10,23 (6,47) UBS Securities Australia Ltd Sell 1,230 (6,74) UBS Securities Australia Ltd Sell 1,230 (7,49) UBS Securities Australia Ltd Sell 1,166 (1,30) UBS Securities Australia Ltd Sell 3,238 (366) UBS Securities Australia Ltd Sell 1,76 (1,198) UBS Sec		ustralia Ltd	Sell	66,328	(7,411)	Ordina
UBS Securities Australia Ltd Sell 50,046 (5,573) UBS Securities Australia Ltd Sell 34,034 (3,790) UBS Securities Australia Ltd Sell 10,189 (1,118) UBS Securities Australia Ltd Sell 10,090 (1,118) UBS Securities Australia Ltd Sell 1,830 (3,790) UBS Securities Australia Ltd Sell 1,830 (3,118) UBS Securities Australia Ltd Sell 1,830 (3,04) UBS Securities Australia Ltd Sell 1,830 (3,04) UBS Securities Australia Ltd Sell 1,180 (3,118) UBS Securities Australia Ltd Sell 1,180 (3,18) UBS Securities Australia Ltd Sell 1,164 (1,19) UBS Securities Australia Ltd Sell 3,283 (3,66) UB	SECTION SECT		Sell	1,042	(116)	Ordina
USS Securities Australia Ltd Buy 63 USS Securities Australia Ltd Sell 101,860 (11,343) UBS Securities Australia Ltd Sell 101,860 (11,343) UBS Securities Australia Ltd Sell 128,782 (14,341) UBS Securities Australia Ltd Sell 128,782 (14,341) UBS Securities Australia Ltd Sell 128,782 (14,341) UBS Securities Australia Ltd Sell 1,287 (14,341) UBS Securities Australia Ltd Sell 1,287 (14,341) UBS Securities Australia Ltd Sell 1,66 (130) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,576 (108) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,577 (129) UBS Securities Australia Ltd Sell 1,577 (138) UBS Securities Australia	REST REST		Sell	50.046	(5.573)	Ordina
UBS Securities Australia Ltd Sell 34,034 (3,790) UBS Securities Australia Ltd Sell 10,1860 (11,343) UBS Securities Australia Ltd Sell 10,1860 (11,343) UBS Securities Australia Ltd Sell 1,28,782 (14,411) UBS Securities Australia Ltd Sell 1,28,782 (30,01) UBS Securities Australia Ltd Sell 2,395 (30,01) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 4,776 (4,929) UBS Securities Australia Ltd Sell 1,156 (13,930) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,776 (1,299) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,776 (1,08) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,776 (1,08)	N		Buy	2	7	Critical
UBS Securities Australia Ltd Sell 101,80 UBS Securities Australia Ltd Sell 101,80 UBS Securities Australia Ltd Sell 101,80 UBS Securities Australia Ltd Sell 128,782 (14,341) UBS Securities Australia Ltd Sell 1,830 (204) UBS Securities Australia Ltd Sell 2,933 (657) UBS Securities Australia Ltd Sell 2,735 (749) UBS Securities Australia Ltd Sell 4,476 (499) UBS Securities Australia Ltd Sell 3,283 (356) UBS Securities Australia Ltd Sell 3,283 (356) UBS Securities Australia Ltd Sell 3,283 (356) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 96 (1) UBS Securities Australia Ltd Sell 96 (1) UBS Securities Australia Ltd Sell 2,225 (2,29) </td <td> N</td> <td></td> <td>Seil</td> <td>34 034</td> <td>(3.790)</td> <td>Critical</td>	N		Seil	34 034	(3.790)	Critical
UBS Securities Australia Ltd Sell 101,860 (11,343) UBS Securities Australia Ltd Sell 596,694 (66,447) UBS Securities Australia Ltd Sell 596,694 (66,447) UBS Securities Australia Ltd Sell 28,000 (3,118) UBS Securities Australia Ltd Sell 1,830 (267) UBS Securities Australia Ltd Sell 6,719 (409) UBS Securities Australia Ltd Sell 6,719 (409) UBS Securities Australia Ltd Sell 7,382 (366) UBS Securities Australia Ltd Sell 1,166 (199) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,157 (1,19) UBS Securities Australia Ltd Sell 2,126 (1,113) UBS Securities Australia Ltd Sell 1,157 (1,108) UBS S			J.	σ	(1)	Cripic
UBS Securities Australia Ltd Sell 596,634 (11,134) UBS Securities Australia Ltd Sell 596,634 (6447) UBS Securities Australia Ltd Sell 1,830 (204) UBS Securities Australia Ltd Sell 28,000 (3,118) UBS Securities Australia Ltd Sell 2,395 (267) UBS Securities Australia Ltd Sell 2,383 (366) UBS Securities Australia Ltd Sell 4,476 (439) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,776 (198) UBS Securities Australia Ltd Sell 2,214 (158) UBS Securities Australia Ltd Sell 2,225 (248) UBS Securities Au			Zell Coll	020 101	ò	5 6
UBS Securities Australia Ltd Sell 1.28,782 (1) UBS Securities Australia Ltd Sell 1.28,782 (14,341) UBS Securities Australia Ltd Sell 1.830 (204) UBS Securities Australia Ltd Sell 1.830 (204) UBS Securities Australia Ltd Sell 2,095 (371) UBS Securities Australia Ltd Sell 4,75 (499) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 2,314 (129) UBS Securities Australia Ltd Sell 2,314 (129) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia L			Sel Sel	701,000	(546,11)	
UBS Securities Australia Ltd Sell 128,782 (14,341) UBS Securities Australia Ltd Sell 2,395 (3,04) UBS Securities Australia Ltd Sell 2,395 (267) UBS Securities Australia Ltd Sell 6,719 (743) UBS Securities Australia Ltd Sell 4,776 (4,331) UBS Securities Australia Ltd Sell 1,782 (31) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,57 (116) UBS Securities Australia Ltd Sell 1,57 (128) UBS Securities Australia Ltd Sell 1,57 (128) UBS Securities Australia Ltd Sell 2,314 (2,68) UBS Securities Australia Ltd Sell 2,125 (1,13) UBS Securities Austral			- L	330,034	(00,447)	Croina
UBS Securities Australia Ltd Sell 1.62.782 (14.34) UBS Securities Australia Ltd Sell 1.830 (204) UBS Securities Australia Ltd Sell 1.830 (204) UBS Securities Australia Ltd Sell 2.395 (267) UBS Securities Australia Ltd Sell 4.75 (749) UBS Securities Australia Ltd Sell 4.73 (360) UBS Securities Australia Ltd Sell 3.283 (366) UBS Securities Australia Ltd Sell 1.576 (108) UBS Securities Australia Ltd Sell 1.576 (108) UBS Securities Australia Ltd Sell 2.225 (248) UBS Securities Australia Ltd Sell 2.126 (11) UBS Securities Australia L			Jac I	2 000	(L)	Craina
UBS Securities Australia Ltd Sell 1,390 (3,11) UBS Securities Australia Ltd Sell 8,073 (304) UBS Securities Australia Ltd Sell 2,395 (305) UBS Securities Australia Ltd Sell 4,476 (499) UBS Securities Australia Ltd Sell 1,166 (130) UBS Securities Australia Ltd Sell 3,283 (365) UBS Securities Australia Ltd Sell 2,314 (128) UBS Securities Australia Ltd Sell 1,157 (1,129) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (13) UBS Securities Australia Ltd Sell 2,126 (13) UBS Securities Australia Ltd			lac lac	787'87	(14,341)	Cldina
UBS Securities Australia Ltd Sell 1,830 (204) UBS Securities Australia Ltd Sell 8,719 (302) UBS Securities Australia Ltd Sell 4,776 (4,83) UBS Securities Australia Ltd Sell 4,776 (130) UBS Securities Australia Ltd Sell 1,166 (130) UBS Securities Australia Ltd Sell 7,382 (823) UBS Securities Australia Ltd Sell 7,382 (823) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,57 (116) UBS Securities Australia Ltd Sell 1,57 (129) UBS Securities Australia Ltd Sell 1,57 (129) UBS Securities Australia Ltd Sell 2,126 (130) UBS Securities Australia Ltd Sell 2,126 (130) UBS Securities Australia Ltd Sell 2,126 (130) UBS Securities Australia Ltd </td <td></td> <td></td> <td>Jac.</td> <td>000,87</td> <td>(3,118)</td> <td>Ordina</td>			Jac.	000,87	(3,118)	Ordina
UBS Securities Australia Ltd Sell 8.073 (500) UBS Securities Australia Ltd Sell 2.395 (267) UBS Securities Australia Ltd Sell 4,776 (499) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (365) UBS Securities Australia Ltd Sell 3,234 (323) UBS Securities Australia Ltd Sell 1,576 (108) UBS Securities Australia Ltd Sell 1,576 (108) UBS Securities Australia Ltd Sell 1,576 (108) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (239) UBS Securities Australia Ltd <td></td> <td></td> <td>Sel.</td> <td>0.58,1</td> <td>(204)</td> <td>Ordina</td>			Sel.	0.58,1	(204)	Ordina
UBS Securities Australia Ltd Sell 6.739 (267) UBS Securities Australia Ltd Sell 4,476 (499) UBS Securities Australia Ltd Sell 4,476 (499) UBS Securities Australia Ltd Sell 1,166 (130) UBS Securities Australia Ltd Sell 45.7 (4023) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (4023) UBS Securities Australia Ltd Sell 3,283 (4023) UBS Securities Australia Ltd Sell 1,157 (118) UBS Securities Australia Ltd Sell 1,157 (118) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (1,23) UBS Securities Australia Ltd Sell 2,126 (1,33) UBS Securities Australia Ltd Sell 2,126 (1,33) UBS Securities Australia Ltd<			Sell	8,073	(006)	Ordina
UBS Securities Australia Ltd Sell 4,719 (749) UBS Securities Australia Ltd Sell 4,776 (499) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 7,382 (823) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,041 (116) UBS Securities Australia Ltd Sell 1,041 (116) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (289) UBS Securities Australia Ltd Sell 2,126 (39) UBS Securities Australia Ltd Sell 2,126 (39) UBS Securities Australia Ltd			Sell	2,395	(267)	Ordina
UBS Securities Australia Ltd Sell 4,476 (499) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (365) UBS Securities Australia Ltd Sell 4,57 (40,23) UBS Securities Australia Ltd Sell 3,283 (365) UBS Securities Australia Ltd Sell 2,314 (258) UBS Securities Australia Ltd Sell 2,314 (258) UBS Securities Australia Ltd Sell 1,176 (108) UBS Securities Australia Ltd Sell 1,177 (129) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell<			Sell	6,719	(749)	Ordina
UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,166 (130) UBS Securities Australia Ltd Sell 7,382 (823) UBS Securities Australia Ltd Sell 7,383 (61) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,57 (116) UBS Securities Australia Ltd Sell 1,57 (116) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,225 (248) UBS Securities Australia Ltd Sell 2,136 (1) UBS Securities Australia Ltd Sell 2,050 (233) UBS Securities Australia Ltd Sell 2,050 (233) UBS Securities Australia Ltd <td< td=""><td></td><td></td><td>Sell</td><td>4,476</td><td>(499)</td><td>Ordina</td></td<>			Sell	4,476	(499)	Ordina
UBS Securities Australia Ltd Sell 1,166 (130) UBS Securities Australia Ltd Sell 7,383 (365) UBS Securities Australia Ltd Sell 457 (513) UBS Securities Australia Ltd Sell 3,283 (365) UBS Securities Australia Ltd Sell 3,234 (323) UBS Securities Australia Ltd Sell 1,776 (108) UBS Securities Australia Ltd Sell 1,776 (108) UBS Securities Australia Ltd Sell 1,57 (108) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (23) UBS Securities Australia Ltd Sell 2,126 (24) UBS Securities Australia Ltd Sell 2,126 (30) UBS Securities Australia Ltd Sell 2,126 (30) UBS Securities Australia Ltd Sell <td></td> <td></td> <td>Seil</td> <td>3,283</td> <td>(398)</td> <td>Ordina</td>			Seil	3,283	(398)	Ordina
UBS Securities Australia Ltd Sell 3,283 (159) UBS Securities Australia Ltd Sell 3,283 (823) UBS Securities Australia Ltd Sell 3,283 (823) UBS Securities Australia Ltd Sell 3,283 (966) UBS Securities Australia Ltd Sell 3,314 (128) UBS Securities Australia Ltd Sell 1,041 (116) UBS Securities Australia Ltd Sell 1,041 (116) UBS Securities Australia Ltd Sell 1,041 (116) UBS Securities Australia Ltd Sell 1,157 (128) UBS Securities Australia Ltd Sell 2,125 (248) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd			199	1166	(120)	i i i
UBS Securities Australia Ltd Sell 7.382 (823) UBS Securities Australia Ltd Sell 457 (81) UBS Securities Australia Ltd Sell 3.283 (366) UBS Securities Australia Ltd Sell 3.283 (366) UBS Securities Australia Ltd Sell 7.314 (108) UBS Securities Australia Ltd Sell 1.041 (116) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (237) UBS Securities Australia Ltd Sell 2,126 (39) UBS Securities Australia Ltd Sell 2,126 (4,391) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell			100	201,1	(00)	old o
UBS Securities Australia Ltd Sell 4,382 (812) UBS Securities Australia Ltd Sell 36,086 (4,023) UBS Securities Australia Ltd Sell 3,183 (366) UBS Securities Australia Ltd Sell 2,314 (228) UBS Securities Australia Ltd Sell 1,776 (108) UBS Securities Australia Ltd Sell 1,157 (129) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (1,29) UBS Securities Australia Ltd Sell 2,126 (200) UBS Securities Australia Ltd Sell 2,126 (200) UBS Securities Australia Ltd Se			la S	3,283	(36b)	Ordina
UBS Securities Australia Ltd Sell 36,086 (4,023) UBS Securities Australia Ltd Sell 3,083 (366) UBS Securities Australia Ltd Sell 3,314 (258) UBS Securities Australia Ltd Sell 1,776 (198) UBS Securities Australia Ltd Sell 1,57 (1,198) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,225 (248) UBS Securities Australia Ltd Sell 2,725 (248) UBS Securities Australia Ltd Sell 2,730 (814) UBS Securities Australia Ltd Sell 2,730 (814) UBS Securities Australia Ltd Sell 2,730 (814) UBS Securities Australia Ltd Sell 2,730 (820) UBS Securities Australia Ltd Se			Sell	/,382	(823)	Ordina
UBS Securities Australia Ltd Sell 3.6,086 (4,023) UBS Securities Australia Ltd Sell 3,283 (366) UBS Securities Australia Ltd Sell 1,776 (108) UBS Securities Australia Ltd Sell 1,776 (118) UBS Securities Australia Ltd Sell 1,157 (198) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,984 (1,13) UBS Securities Australia Ltd <t< td=""><td></td><td></td><td>ax.</td><td>457</td><td>(51)</td><td>Ordina</td></t<>			ax.	457	(51)	Ordina
UBS Securities Australia Ltd Sell 3.283 (366) UBS Securities Australia Ltd Sell 2,314 (258) UBS Securities Australia Ltd Sell 1,776 (198) UBS Securities Australia Ltd Sell 1,041 (116) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,225 (248) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1,13) UBS Securities Australia Ltd Sell 2,109 (1,13) UBS Securities Australia Ltd Sell 2,109 (233) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 2,090 (230) UBS Securities Australia Ltd Sell </td <td></td> <td></td> <td>Sell</td> <td>36,086</td> <td>(4,023)</td> <td>Ordina</td>			Sell	36,086	(4,023)	Ordina
UBS Securities Australia Ltd Sell 2,314 (728) UBS Securities Australia Ltd Sell 1,776 (108) UBS Securities Australia Ltd Sell 1,041 (116) UBS Securities Australia Ltd Sell 9 (1,1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,225 (248) UBS Securities Australia Ltd Sell 2,126 (27) UBS Securities Australia Ltd Sell 2,126 (37) UBS Securities Australia Ltd Sell 2,126 (39) UBS Securities Australia Ltd Sell 2,020 (233) UBS Securities Australia Ltd Sell 2,030 (390) UBS Securities Australia Ltd Sell 2,040 (200) UBS Securities Australia Ltd Sell 2,040 (200) UBS Securities Australia Ltd Sell <td></td> <td></td> <td>Sell</td> <td>3,283</td> <td>(396)</td> <td>Ordina</td>			Sell	3,283	(396)	Ordina
UBS Securities Australia Ltd Sell 969 (108) UBS Securities Australia Ltd Sell 1,776 (198) UBS Securities Australia Ltd Sell 1,157 (129) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 9,984 (1,13) UBS Securities Australia Ltd Sell 2,126 (13) UBS Securities Australia Ltd Sell 9,984 (1,13) UBS Securities Australia Ltd Sell 2,126 (1,29) UBS Securities Australia Ltd Sell 2,090 (2,23) UBS Securities Australia Ltd Sell 2,090 (2,29) UBS Securities Australia Ltd Sell 2,170 (2,29) UBS Securities Australia Ltd Sell 2,170 (2,60) UBS Securities Australia Ltd <t< td=""><td></td><td></td><td>Sell</td><td>2,314</td><td>(258)</td><td>Ordina</td></t<>			Sell	2,314	(258)	Ordina
UBS Securities Australia Ltd Sell 1,776 (198) UBS Securities Australia Ltd Sell 1,041 (116) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,125 (248) UBS Securities Australia Ltd Sell 2,125 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,130 (1,13) UBS Securities Australia Ltd Sell 2,109 (233) UBS Securities Australia Ltd Sell 2,117 (290) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 1,790 (200) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell			Sell	696	(108)	Ordina
UBS Securities Australia Ltd Sell 1,157 (115) UBS Securities Australia Ltd Sell 1,157 (129) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,725 (248) UBS Securities Australia Ltd Sell 2,726 (1) UBS Securities Australia Ltd Sell 9,984 (1,113) UBS Securities Australia Ltd Sell 9,984 (1,113) UBS Securities Australia Ltd Sell 2,020 (237) UBS Securities Australia Ltd Sell 2,030 (233) UBS Securities Australia Ltd Sell 2,030 (250) UBS Securities Australia Ltd Sell 2,117 (250) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd <t< td=""><td> </td><td></td><td>Seil</td><td>1 776</td><td>(198)</td><td>orio C</td></t<>	 		Seil	1 776	(198)	orio C
UBS Securities Australia Ltd Sell 1,197 (179) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,225 (248) UBS Securities Australia Ltd Sell 2,126 (1,13) UBS Securities Australia Ltd Sell 2,126 (1,13) UBS Securities Australia Ltd Sell 2,090 (1,13) UBS Securities Australia Ltd Sell 2,090 (239) UBS Securities Australia Ltd Sell 2,090 (239) UBS Securities Australia Ltd Sell 2,090 (239) UBS Securities Australia Ltd Sell 1,17 (200) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd <t< td=""><td> </td><td></td><td>los J</td><td>1041</td><td>(110)</td><td></td></t<>	 		los J	1041	(110)	
UBS Securities Australia Ltd Sell 1, 37 (1, 2) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,225 (2,48) UBS Securities Australia Ltd Sell 2,125 (1) UBS Securities Australia Ltd Sell 2,136 (1,13) UBS Securities Australia Ltd Sell 4,769 (1,13) UBS Securities Australia Ltd Sell 2,090 (2,33) UBS Securities Australia Ltd Sell 2,090 (2,33) UBS Securities Australia Ltd Sell 2,090 (2,09) UBS Securities Australia Ltd Sell 1,76 (2,00) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd	1 		Jac Jac Jac Jac Jac Jac Jac Jac Jac Jac	1,04	(Q 100 S	Craina
UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,225 (248) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 9,984 (1,13) UBS Securities Australia Ltd Sell 2,984 (1,13) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 2,090 (239) UBS Securities Australia Ltd Sell 2,170 (209) UBS Securities Australia Ltd Sell 2,177 (200) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd			nac nac	1,137	(671)	Ordina
UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2.225 (248) UBS Securities Australia Ltd Sell 2.126 (1) UBS Securities Australia Ltd Sell 2.126 (1) UBS Securities Australia Ltd Sell 2.050 (1).13 UBS Securities Australia Ltd Sell 4.7759 (3.23) UBS Securities Australia Ltd Sell 2.090 (2.33) UBS Securities Australia Ltd Sell 2.090 (2.39) UBS Securities Australia Ltd Sell 7.79 (2.00) UBS Securities Australia Ltd Sell 4.48S (500) UBS Securities Australia Ltd			Sell	on l	Ē	Ordina
UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,225 (248) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 9,984 (1,113) UBS Securities Australia Ltd Sell 7,302 (814) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 1,79 (200) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 10,558 (1,777) UBS Securities Australia Ltd Sell (1,000) (1,000)			Sell	50	Ē	Ordina
UBS Securities Australia Ltd Sell 2 (1) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1,137) UBS Securities Australia Ltd Sell 9,984 (1,113) UBS Securities Australia Ltd Sell 4,4769 (4,911) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 2,117 (236) UBS Securities Australia Ltd Sell 2,117 (200) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 10,558 (1,000)			Sell	6	Ê	Ordina
UBS Securities Australia Ltd Sell 2.255 (248) UBS Securities Australia Ltd Sell 2,126 (1) UBS Securities Australia Ltd Sell 2,126 (1,133) UBS Securities Australia Ltd Sell 2,094 (1,113) UBS Securities Australia Ltd Sell 4,750 (4,391) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 2,090 (239) UBS Securities Australia Ltd Sell 1,17 (200) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell (1,000)			Sell	6	(1)	Ordina
UBS Securities Australia Ltd Sell 9 (1) USS Securities Australia Ltd Sell 2,126 (237) USS Securities Australia Ltd Sell 9,984 (1,113) USS Securities Australia Ltd Sell 44,769 (4,991) USS Securities Australia Ltd Sell 2,090 (233) USS Securities Australia Ltd Sell 2,117 (236) USS Securities Australia Ltd Sell 2,172 (590) USS Securities Australia Ltd Sell 4,485 (500) USS Securities Australia Ltd Sell 4,485 (500) USS Securities Australia Ltd Sell 4,485 (500) USS Securities Australia Ltd Sell 10,588 (1,777) USS Securities Australia Ltd Sell 10,588 (1,700)			Sell	2,225	(248)	Ordina
UBS Securities Australia Ltd Sell 2,126 (377) UBS Securities Australia Ltd Sell 7,302 (81,113) UBS Securities Australia Ltd Sell 7,302 (81,41) UBS Securities Australia Ltd Sell 44,769 (4,991) UBS Securities Australia Ltd Sell 2,117 (238) UBS Securities Australia Ltd Sell 2,117 (239) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (1,000)			Sell	6	(i)	Ordina
UBS Securities Australia Ltd Sell 9,984 (1,113) UBS Securities Australia Ltd Sell 4,792 (814) UBS Securities Australia Ltd Sell 4,769 (233) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 5,292 (590) UBS Securities Australia Ltd Sell 1,79 (20) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 10,558 (1,777) UBS Securities Australia Ltd Sell 10,558 (1,700)	Ť		Sell	2.126	(237)	Ordina
UBS Securities Australia Ltd Sell 7,302 (814) UBS Securities Australia Ltd Sell 4,769 (4,991) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 2,117 (396) UBS Securities Australia Ltd Sell 1,79 (20) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 1,0558 (1,777) UBS Securities Australia Ltd Sell (1,000)	Ť		J	0 984	(1 113)	cipio
UBS Securities Australia Ltd Sell 44,769 (931) UBS Securities Australia Ltd Sell 2,090 (233) UBS Securities Australia Ltd Sell 2,117 (236) UBS Securities Australia Ltd Sell 1,79 (20) UBS Securities Australia Ltd Sell 1,79 (20) UBS Securities Australia Ltd Sell 4,485 (50)			Sell	7 303	(2,10)	
UBS Securities Australia Ltd Sell 44,769 (4,591) UBS Securities Australia Ltd Sell 2,117 (236) UBS Securities Australia Ltd Sell 2,117 (236) UBS Securities Australia Ltd Sell 1,29 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 10,558 (1,000) UBS Securities Australia Ltd Sell 10,558 (1,000)	۳		201	205,1	(014)	Ordina
UBS Securities Australia Ltd Sell 2,190 (233) UBS Securities Australia Ltd Sell 2,117 (390) UBS Securities Australia Ltd Sell 1,79 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 1,0558 (1,777) UBS Securities Australia Ltd Sell 8,970 (1,000)	7		Seil	44,769	(4,991)	Ordina
UBS Securities Australia Ltd Sell 2,117 (236) UBS Securities Australia Ltd Sell 5,292 (590) UBS Securities Australia Ltd Sell 179 (20) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 10,558 (1,777) UBS Securities Australia Ltd Sell 8,970 (1,000)	7		Seil	2,090	(233)	Ordina
UBS Securities Australia Ltd Sell 5,292 (590) UBS Securities Australia Ltd Sell 179 (20) UBS Securities Australia Ltd Sell 4,485 (50) UBS Securities Australia Ltd Sell 4,485 (50) UBS Securities Australia Ltd Sell 10,558 (1,177) UBS Securities Australia Ltd Sell 8,970 (1,000)	╛		Sell	2,117	(236)	Ordina
UBS Securities Australia Ltd Sell 179 (20) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 1,0558 (1,777) UBS Securities Australia Ltd Sell 8,970 (1,000)	i		Seli	5,292	(065)	Ordina
UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 1,780 (1,700) UBS Securities Australia Ltd Sell 10,558 (1,700) UBS Securities Australia Ltd Sell 8,970 (1,000)	Ī		Seli	179	(20)	Ordina
UBS Securities Australia Ltd Sell 4,485 (500) UBS Securities Australia Ltd Sell 10,558 (1,177) UBS Securities Australia Ltd Sell 8,970 (1,000)	SBO		Seli	4,485	(200)	Ordina
UBS Securities Australia Ltd Sell 10,558 (1,177) UBS Securities Australia Ltd Sell 8,970 (1,000)	۲		Sell	4.485	(200)	Ordinar
UBS Securities Australia Ltd Seil 8,970 (1,000)	f		le S	10.558	(4, 177)	region
Obs securities Australia Ltd Sell 8,970 (1,000)	Т		120	800,01	(// ')	200
	T		Sell	0/8/0	(1,000)	Ordinar

05-May-14	I IBC AG London Branch	Coll	3/	/5/	i i i
05-May-14	LIRS AG London Branch	Cell	243	(c)	Ordinary
05-Mav-14	UBS AG tondon Branch	3	791	(00)	
05-May-14	UBS AG london Branch	والإ	2 966	(331)	
05-May-14	UBS AG London Branch	وا	6 757		Ordinary
05-Mav-14	I IBS AG London Branch	Cell	1 103		
OF May 14	TOTAL CONTROL OF SOLUTION OF S	201	001,1	(132)	Ordinary
05 Mar. 14	UBS AG LOTIGOT Branch) sell	796,1		Ordinary
th-yelvi-co	UBS AG LOTIGOT Blarict	Sell	507'5	(/8c)	Ordinary
03-Mdy-14	UBS AG LONGON Branch	Sell	797'		Ordinary
05-May-14	UBS AG London Branch	Sell	789	(88)	Ordinary
US-May-14	UBS AG London Branch	Sell	1,311		Ordinary
05-May-14	UBS AG London Branch	e	5,945	Ĭ	Ordinary
05-May-14	UBS AG London Branch	Sell	224	(25)	Ordinary
05-May-14	UBS AG London Branch	Sell.	619		Ordinary
05-May-14	UBS AG London Branch	Se!	2,413		Ordinary
05-May-14	UBS AG London Branch	Sell	5,705	(989)	Ordinary
05-May-14	UBS AG London Branch	Sell	3,992		Ordinary
05-May-14	UBS AG London Branch	Sell	5,346		Ordinary
05-May-14	UBS AG tondon Branch	Sell	2,260	(252)	Ordinary
05-May-14	UBS AG London Branch	Sell	2,727		Ordinary
05-May-14	UBS AG London Branch	Sell	4,341		Ordinary
05-May-14	UBS AG London Branch	Seli	1,686	(188)	Ordinary
05-May-14	UBS AG London Branch	Sell	18	(2)	Ordinary
05-May-14	UBS AG London Branch	Sell	18	(2)	Ordinary
05-May-14	UBS AG London Branch	Sell		(231)	Ordinary
05-May-14	UBS AG London Branch	Sell	10,396	(1,159)	Ordinary
05-May-14	UBS AG London Branch	Sei	206	(23)	Ordinary
05-May-14	UBS AG London Branch	Sell	323	(36)	Ordinary
05-May-14	UBS AG London Branch	Sell	215	(24)	Ordinary
05-May-14	UBS AG London Branch	Sell	421	(47)	Ordinary
05-May-14	UBS AG London Branch	Sell	18	(2)	Ordinary
05-May-14	UBS AG London Branch	Sell	36	(4)	Ordinary
05-May-14	UBS AG London Branch	Sell	10,573	(1,180)	Ordinary
05-May-14	UBS AG London Branch	Seli	σ.	(1)	Ordinary
05-May-14	UBS AG London Branch	Sell	2,195	(245)	Ordinary
U5-May-14	UBS AG London Branch	Sell	5,779	(645)	Ordinary
US-IMay-14	UBS AG London Branch	Jell	11,684	(1,304)	Ordinary
US-May-14	UBS AG London Branch	Sell	358	(40)	Ordinary
05-May-14	UBS AG London Branch	Sell	304	(34)	Ordinary
05-ividy-14	UBS AG London Branch	Sell	3,5/5	(665)	Ordinary
OF 14	UBS AG London Branch	Jac.	6	(i)	Ordinary
OF 14	UBS AG London Branch	Sell	856,5	(440)	Ordinary
US-IVIAY-14	UBS AG London Branch	les.	046,8	(1,000)	Ordinary
05-May-14	UBS AG London Branch	ll sell	1,181	(132)	Ordinary
05-May-14	UBS AG London Branch	Sell	3,025	(338)	Ordinary
05-May-14	UBS AG London Branch	Sell	9,487	(1,060)	Ordinary
05-May-14	UBS AG London Branch	Self	86	(11)	Ordinary
05-May-14	UBS AG London Branch	Sell	6	(1)	Ordinary
05-May-14	UBS AG London Branch	Sell	10,651	(1,190)	Ordinary
05-May-14	UBS AG London Branch	Sell	2,363	(564)	Ordinary
05-May-14	UBS AG London Branch	Sell	1,378	(154)	Ordinary
05-May-14	UBS AG London Branch	Sell	1,333	(149)	Ordinary
05-May-14	UBS AG London Branch	Set	447	(20)	Ordinary
05-May-14	UBS AG London Branch	Sell	1,602	(179)	Ordinary
05-May-14	UBS AG London Branch	Sell Sell	8,834	(282)	Ordinary
05-May-14	UBS AG London Branch	Sell	8,950	(1,000)	Ordinary
05-May-14	UBS AG tondon Branch	Sell	54,595	(6,100)	Ordinary
05-May-14	UBS AG London Branch	Sell	358	(40)	Ordinary
US-May-14	UBS AG London Branch	Seil	618	(69)	Ordinary
US-May-14	UBS Securities Australia Ltd	gnà	695'7	/87	Ordinary
US-May-14	UBS Securities Australia Ltd	Sell	36,198	(4,040)	Ordinary
U5-May-14	UBS Securities Australia Ltd	Buy	392,222	43,810	Ordinary
05-May-14	UBS Securities Australia Ltd	Zell Sell	226'692	(30,135)	Ordinary
05-May-14	UBS Securities Australia Ltd	Buy	44,900	5,000	Ordinary
05-May-14	UBS Securities Australia Ltd	Sell	22,400	(2,500)	Ordinary
05-May-14	UBS Securities Australia Ltd	Buy	29,568	3,300	Ordinary
05-May-14	UBS Securities Australia Ltd	Buy	15,698	1,754	Ordinary

URS Actionate Barach Sell 1,749 (15) URS Actionate Barach Sell 1,749 (15) URS Actionate Barach Sell 1,749 (15) URS Actionate Barach Sell 1,749 (16) URS Actionate Barach Sell 1,729 (16) URS Actionate Barach Sell 641 (72) URS Actionate Barach Sell 641 (72) URS Actionate Barach Sell 641 (72) URS Actionate Barach Sell 1,729 (14) URS Actionate Barach Sell 2,729 (20) URS Actionate Barach Sell <th>s securities Australia Ltd 5 Securities Australia Ltd 5 AG London Branch</th> <th>lles</th> <th>7,849</th> <th>(875) (2)</th> <th>Ordinary</th>	s securities Australia Ltd 5 Securities Australia Ltd 5 AG London Branch	lles	7,849	(875) (2)	Ordinary
URS ACC LONDER BRANCH Sell 1,834 (87) URS ACC LONDER BRANCH Sell 1,834 (87) URS ACC LONDER BRANCH Sell 1,672 (18) URS ACC LONDER BRANCH Sell 1,672 (19) URS ACC LONDER BRANCH Sell 614 (17) URS ACC LONDER BRANCH Sell 614 (17) URS ACC LONDER BRANCH Sell 614 (17) URS ACC LONDER BRANCH Sell 1,729 (18) URS ACC LONDER BRANCH Sell 1,729 (19) URS ACC LONDER BRANCH Sell 1,729 (19) URS ACC LONDER BRANCH Sell 1,729 (10) URS ACC LONDER BRANCH Sell 1,729 (10) URS ACC LONDER BRANCH Sell 2,720 (10) URS ACC LONDER BRANCH Sell 2,720 (10) URS ACC LONDER BRANCH Sell 2,722 (10) URS ACC LONDER BRANCH Sell 2,722 (10) URS ACC LONDER BRANCH	s acturities Australia Ltd s AG London Branch	tlac	1,849	(2)	Ordinary
URS ACL LONDON Branch Sell 12.2 (2.5) URS ACL LONDON Branch Sell 1.0 (2.5) (3.5) URS ACL LONDON Branch Sell 64.1 (2.5) (3.5) URS ACL LONDON Branch Sell 64.1 (7.2) (8.5) URS ACL LONDON Branch Sell 64.1 (7.2) (8.5) URS ACL LONDON Branch Sell 7.2 (8.5) (1.6) URS ACL LONDON Branch Sell 7.2 (8.1) (8.5) URS ACL LONDON Branch Sell 7.2 (8.1) (8.5) URS ACL LONDON Branch Sell 7.2 (8.5) (1.6) URS ACL LONDON Branch Sell 7.2 (8.5) (1.0) URS ACL LONDON Branch Sell 7.2 (8.5) (1.6) URS ACL LONDON Branch Sell 7.2 (8.5) (1.7) URS ACL LONDON Branch Sell 7.2 (8.5) (1.6) URS ACL LONDON Branch Sell 7.2 (8.5) (1.6) <	AG LORIGOT BIRDING		<u>80</u>	(7)	10000
UBS ACL LONDON Branch Sell 2.73 (2.9) UBS ACL LONDON Branch Sell 61.4 (1.9) UBS ACL LONDON Branch Sell 61.4 (1.9) UBS ACL LONDON Branch Sell 61.4 (1.9) UBS ACL LONDON Branch Sell 64.0 (1.9) UBS ACL LONDON Branch Sell 7.20 (8.1) UBS ACL LONDON Branch Sell 4.40 (2.9) UBS ACL LONDON Branch Sell 4.40 (2.9) UBS ACL LONDON Branch Sell 1.725 (1.10) UBS ACL LONDON Branch Sell 7.72 (2.9) UBS ACL LONDON Branch		liac II	-		Officially
URS ACL London Branch Sell 160 (18) URS ACL London Branch Sell 64.1 (72) URS ACL London Branch Sell 64.1 (72) URS ACL London Branch Sell 64.1 (72) URS ACL London Branch Sell 64.0 (4.95) URS ACL London Branch Sell 7.2 (8.1) URS ACL London Branch Sell 7.2 (8.2) URS ACL London Branch <th< td=""><td>AG London Branch</td><td>Sell</td><td>223</td><td>(25)</td><td>Ordinary</td></th<>	AG London Branch	Sell	223	(25)	Ordinary
USA AG London Branch Sel 614 619 619 619 618 614	s AG London Branch	Sell	27	(3)	Ordinary
UBS ACL London Branch Sell 611 (172) UBS ACL London Branch Sell 4.40 (1.72) UBS ACL London Branch Sell 1,783 (1.93) UBS ACL London Branch Sell 1,783 (1.93) UBS ACL London Branch Sell 1,783 (1.93) UBS ACL London Branch Sell 4,074 (4.31) UBS ACL London Branch Sell 4,074 (4.31) UBS ACL London Branch Sell 7,03 (1.93) UBS ACL London Branch Sell 7,22 (6.10) UBS ACL London Branch Sell 5,22 (6.10) UBS ACL London Branch Sell 7,22 (6.10) UBS ACL London Branch Sell 7,23 (6.10) UBS ACL London Branch Sell 7,23 (6.10) UBS ACL London Branch Sell 7,24 (6.10) UBS ACL London Branch Sell 7,24 (6.10) UBS ACL London Branch Sell 7,24 (6.20) UBS	AG London Branch	Sell	160	(18)	Ordinary
Libis Act London Barach Sell 64.1 (7.9) Libis Act London Barach Sell 4.09 (6.9) Libis Act London Barach Sell 1,233 (14.1) Libis Act London Barach Sell 1,234 (14.1) Libis Act London Barach Sell 1,230 (1.0) Libis Act London Barach Sell 6.00 (7.2) Libis Act London Barach Sell 1,240 (1.0) Libis Act London Barach Sell 1,245 (1.0) Libis Act London Barach Sell 1,245 (1.0) Libis Act London Barach Sell 1,245 (1.6) Libis Act London Barach Sell 1,245 (1.6) Libis Act London Barach Sell 2,241 (2.6) Libis Act London Barach Sell 2,241 (2.6)	s AG London Branch		614	(69)	Ordinary
USS AG London Barach Seil 7,090 (696) USS AG London Barach Seil 7,090 (697) USS AG London Barach Seil 1,293 (741) USS AG London Barach Seil 4,014 (753) USS AG London Barach Seil 4,014 (753) USS AG London Barach Seil 4,014 (753) USS AG London Barach Seil 7,000 (753) USS AG London Barach Seil	s AG London Branch	lles	28	(72)	Ordinary
USS AG London Branch Sell 1,253 (1,99) USS AG London Branch Sell 1,253 (1,99) USS AG London Branch Sell 1,263 (1,99) USS AG London Branch Sell 1,263 (1,99) USS AG London Branch Sell 1,263 (1,10) USS AG London Branch Sell 1,263 (1,10) USS AG London Branch Sell 5,209 (610) USS AG London Branch Sell 7,209 (610) USS AG London Branch Sell 7,209 (620) USS AG London Branch Sell 7,209 (620) USS AG London Branch Sell 7,209 (620) USS AG London Branch Sell 7,209 (621) USS AG London Bra	AG London Branch	روا	0.000	(406)	1000
Bis Act London Branch Sell 1,789 (199) Loss Act London Branch Sell 1,233 (141) Loss Act London Branch Sell 4,014 (143) Bis Act London Branch Sell 7,09 (79) List Sack London Branch Sell 7,09 (79) List Sack London Branch Sell 7,29 (70) List Sack London Branch Sell 6,209 (60) List Sack London Branch Sell 2,20 (71) List Sack London Branch Sell 2,20 (73) List Sack London Branch Sell 2,20 (73) List Sack London Branch Sell 2,20 (73) List Sack London Branch Sell 2,71 2 List Sack London Branch Sell 2,05 (65) List Sack London Branch Sell 2,05 (73) List Sack London Branch Sell 2,05 (73) List Sack London Branch Sell 2,07 (73) List Sack	AG London Branch	les	720	(101)	, delipio
USS AG London Barch Sell 1,253 (149) USS AG London Barch Sell 1,233 (149) USS AG London Barch Sell 1,233 (149) USS AG London Barch Sell 1,234 (170) USS AG London Barch Sell 1,234 (170) USS AG London Barch Sell 5,429 (170) USS AG London Barch Sell 5,429 (171) USS AG London Barch Sell 5,429 (170) USS AG London Barch Sell 5,429 (171) USS AG London Barch Sell 5,429 (171) USS AG London Barch Sell 5,429 (171) USS AG London Barch Sell 1,256 (160) USS AG London Barch Sell 1,258 (161) USS AG London Barch Sell 1,258 (162) USS AG London Barch Sell 1,270 (28) USS AG London Barch Sell 1,270 (28) USS AG London Barch Sell	AG London Branch	100	0111	(10)	, iii
UBS ACE LONDON BRANCH SMIII 4,014 3 (451) UBS ACE LONDON BRANCH SMIII 4,014 3 (451) UBS ACE LONDON BRANCH SMIII 1,949 (279) UBS ACE LONDON BRANCH SMIII 6,029 (710) UBS ACE LONDON BRANCH SMIII 5,429 (710) UBS ACE LONDON BRANCH SMIII 2,540 (710) UBS ACE LONDON BRANCH SMIII 1,246 (740) UBS ACE LONDON BRANCH SMIII 2,741 (852) UBS ACE LONDON BRANCH SMIII 2,741 (854) UBS ACE LONDON BRANCH SMIII 2,741 (854) UBS ACE LONDON BRANCH SMIII 2,741 (877) UBS ACE LONDON BRANCH SMIII 2,741 (877) UBS ACE LONDON BRANCH SMIII 2,741 (877)	A CONTOUR DISTRICT	nac .	60/,1	(661)	Ordinary
UBS AG London Branch Sell 4,014 (45) UBS AG London Branch Sell 1,026 (72) UBS AG London Branch Sell 7,02 (72) UBS AG London Branch Sell 64,0 (72) UBS AG London Branch Sell 6,209 (610) UBS AG London Branch Sell 6,209 (610) UBS AG London Branch Sell 6,209 (610) UBS AG London Branch Sell 1,246 (140) UBS AG London Branch Sell 1,246 (140) UBS AG London Branch Sell 1,246 (140) UBS AG London Branch Sell 2,075 (65) UBS AG London Branch Sell 2,075 (62) UBS AG London Branch Sell 2,075 (62) UBS AG London Branch <	AG London Branch	Sell	1,253	(141)	Ordinary
UBS AG London Branch Sell 1,949 (71) UBS AG London Branch Sell 703 (72) UBS AG London Branch Sell 620 (710) UBS AG London Branch Sell 6,429 (610) UBS AG London Branch Sell 6,229 (610) UBS AG London Branch Sell 6,229 (610) UBS AG London Branch Sell 7,204 (73) UBS AG London Branch Sell 7,204 (740) UBS AG London Branch Sell 7,204 (75) UBS AG London Branch	s AG London Branch	Sell	4,014	(451)	Ordinary
UBS AG London Barrch Sell 703 (79) UBS AG London Barrch Sell 640 (72) UBS AG London Barrch Sell 640 (72) UBS AG London Barrch Sell 6,209 (110) UBS AG London Barrch Sell 6,209 (110) UBS AG London Barrch Sell 6,209 (110) UBS AG London Barrch Sell 1,246 (173) UBS AG London Barrch Sell 1,246 (173) UBS AG London Barrch Sell 3,204 (180) UBS AG London Barrch Sell 3,204 (180) UBS AG London Barrch Sell 7,128 (173) UBS AG London Barrch Sell 7,128 (140) UBS AG London Barrch Sell 7,128 (140) UBS AG London Barrch Sell 7,128 (173) UBS AG London Barrch Sell 7,128 (174) UBS AG London Barrch Sell 7,128 (174) UBS AG London Barrch	AG London Branch	Sell	1,949	(219)	Ordinary
UBS AG London Barch SSI T/18 (20) UBS AG London Barch Sell 5.429 (110) UBS AG London Barch Sell 5.429 (110) UBS AG London Barch Sell 5.279 (110) UBS AG London Barch Sell 6.209 (650) UBS AG London Barch Sell 5.27 (51) UBS AG London Barch Sell 3.204 (161) UBS AG London Barch Sell 3.204 (162) UBS AG London Barch Sell 3.204 (163) UBS AG London Barch Sell 3.204 (20) UBS AG London Barch Sell 3.204 (37) UBS AG London Barch Sell 3.204 (37) UBS AG London Barch Sell	5 AG London Branch	Sell	703	(62)	Ordinary
URS AG London Branch Sell 640 (72) UBS AG London Branch Sell 6,209 (610) UBS AG London Branch Sell 6,209 (610) UBS AG London Branch Sell 6,209 (658) UBS AG London Branch Sell 1,246 (13) UBS AG London Branch Sell 3,204 (658) UBS AG London Branch Sell 3,204 (659) UBS AG London Branch Sell 3,204 (651) UBS AG London Branch Sell 3,204 (657) UBS AG London Branch Sell 7,201 (857) UBS AG London Branch Sell 7,201 (857) UBS AG London Branch Sell 7,201 (857) UBS AG London Branch Sell 7,201 (877) UBS AG London Branch Sell 7,201 (877) UBS AG London Branch Sell 7,202 (871) UBS AG London Branch Sell 7,202 (872) UBS AG London Branch	AG London Branch	les	178	(00)	Ordinan
USS AG London Branch Sell 5729 (17) USS AG London Branch Sell 5,429 (10) USS AG London Branch Sell 6,209 (610) USS AG London Branch Sell 1,540 (173) USS AG London Branch Sell 267 (39) USS AG London Branch Sell 1,243 (19) USS AG London Branch Sell 1,254 (140) USS AG London Branch Sell 1,245 (161) USS AG London Branch Sell 1,128 (165) USS AG London Branch Sell 1,128 (167) USS AG London Branch Sell 1,207 (151) USS AG London Branch Sell 1,207 (151) USS AG London Branch Sell 1,207 (24) USS AG London Branch Sell 1,207 (27) USS AG London Branch Sell 1,207 (27) USS AG London Branch Sell 1,207 (27) USS AG London Branch	AG London Branch	Je J	079	(02)	, in a
URS AG London Branch Sell 5,479 (110) URS AG London Branch Sell 6,209 (698) URS AG London Branch Sell 1,204 (630) URS AG London Branch Sell 1,246 (140) URS AG London Branch Sell 2,925 (651) URS AG London Branch Sell 2,031 (854) URS AG London Branch Sell 2,032 (61) URS AG London Branch Sell 2,031 (854) URS AG London Branch Sell 2,031 (854) URS AG London Branch Sell 2,031 (871) URS AG London Branch Sell 2,031 (872) URS AG London Branch Sell 2,032 (437) URS AG London Branch <td>ACTORDO DESCRIP</td> <td>114</td> <td>210</td> <td>77.7</td> <td>Cruinary</td>	ACTORDO DESCRIP	114	210	77.7	Cruinary
USA SG London Branch Sell 5,429 (690) USB AG London Branch Sell 6,249 (691) USB AG London Branch Sell 1,540 (173) USB AG London Branch Sell 1,540 (173) USB AG London Branch Sell 1,640 (173) USB AG London Branch Sell 1,245 (140) USB AG London Branch Sell 1,245 (161) USB AG London Branch Sell 1,245 (165) USB AG London Branch Sell 2,925 (165) USB AG London Branch Sell 2,925 (165) USB AG London Branch Sell 2,925 (167) USB AG London Branch Sell 2,925 (170) USB AG London Branch Sell 2,925 (170) USB AG London Branch Sell 1,781 (271) USB AG London Branch Sell 1,781 (437) USB AG London Branch Sell 1,781 (437) USB AG London Branch <td>A C LONGOI DIGILATI</td> <td>Jen .</td> <td>6/6</td> <td>6</td> <td>Ordinary</td>	A C LONGOI DIGILATI	Jen .	6/6	6	Ordinary
UBS AG London Branch Sell 6,209 (139) UBS AG London Branch Sell 1,500 (133) UBS AG London Branch Sell 267 (33) UBS AG London Branch Sell 1,246 (140) UBS AG London Branch Sell 1,246 (140) UBS AG London Branch Sell 1,246 (140) UBS AG London Branch Sell 1,246 (151) UBS AG London Branch Sell 2,075 (163) UBS AG London Branch Sell 2,075 (163) UBS AG London Branch Sell 2,075 (163) UBS AG London Branch Sell 2,075 (170) UBS AG London Branch Sell 1,977 (180) UBS AG London Branch Sell 1,977 (170) UBS AG London Branch Sell 1,977 (170) UBS AG London Branch Sell 1,977 (170) UBS AG London Branch Sell 2,973 (170) UBS AG London Branch	AG London Branch	Self	5,429	(610)	Ordinary
UBS AG London Branch Sell 1,540 (173) UBS AG London Branch Sell 267 (39) UBS AG London Branch Sell 3,204 (30) UBS AG London Branch Sell 1,455 (140) UBS AG London Branch Sell 1,455 (161) UBS AG London Branch Sell 7,128 (800) UBS AG London Branch Sell 7,128 (800) UBS AG London Branch Sell 7,128 (800) UBS AG London Branch Sell 2,14 (24) UBS AG London Branch Sell 2,165 (57) UBS AG London Branch Sell 7,07 (81) UBS AG London Branch Sell 7,07 (87) UBS AG London Branch Sell 2,541 (24) UBS AG London Branch Sell 2,541 (27) UBS AG London Branch Sell 2,541 (27) UBS AG London Branch Sell 2,541 (27) UBS AG London Branch	AG London Branch	Seli	6,209	(869)	Ordinary
UBS AC London Branch Sell 472 (53) UBS AC London Branch Seal 3.24 (30) UBS AC London Branch Seal 3.24 (360) UBS AC London Branch Seal 3.24 (360) UBS AC London Branch Seal 1.246 (141) UBS AC London Branch Seal 7.24 (360) UBS AC London Branch Seal 7.24 (360) UBS AC London Branch Seal 7.24 (15) UBS AC London Branch Seal 7.07 (28) UBS AC London Branch Seal 7.24 (36) UBS AC London Branch Seal 7.24 (37) UBS AC London Branch Seal 7.24 (27) UBS AC London Branch Seal 7.24 (27) UBS AC London Branch Seal	5 AG London Branch	Sell	1,540	(173)	Ordinary
UBS AG London Banch Sell 267 (30) UBS AG London Banch Sell 3.204 (35) UBS AG London Banch Sell 1.243 (161) UBS AG London Banch Sell 1.243 (161) UBS AG London Banch Sell 7.28 (80) UBS AG London Banch Sell 7.20 (84) UBS AG London Banch Sell 7.201 (84) UBS AG London Banch Sell 7.201 (84) UBS AG London Banch Sell 7.201 (85) UBS AG London Banch Sell 7.201 (87) UBS AG London Banch Sell 7.201 (87) UBS AG London Banch Sell 7.201 (73) UBS AG London Banch Sell <t< td=""><td>AG London Branch</td><td>Sell</td><td>472</td><td>(53)</td><td>Ordinary</td></t<>	AG London Branch	Sell	472	(53)	Ordinary
URS AG London Branch Seal 3,244 (36) URS AG London Branch Seal 1,246 (140) URS AG London Branch Seal 1,246 (140) URS AG London Branch Seal 1,246 (140) URS AG London Branch Seal 2,435 (151) URS AG London Branch Seal 7,128 (850) URS AG London Branch Seal 7,075 (85) URS AG London Branch Seal 7,075 (871) URS AG London Branch Seal 7,075 (872) URS AG London Branch Seal 7,074 (872) URS AG London Branch Seal 7,449 (872) URS AG London Branch	AG London Branch	9	735	(00)	i i i
UBS AG London Banch Sell 1,244 (140) UBS AG London Banch Sell 1,244 (140) UBS AG London Banch Sell 1,245 (161) UBS AG London Banch Sell 7,28 (165) UBS AG London Banch Sell 7,28 (166) UBS AG London Banch Sell 7,01 (854) UBS AG London Banch Sell 7,01 (854) UBS AG London Banch Sell 7,01 (854) UBS AG London Banch Sell 7,07 (233) UBS AG London Banch Sell 7,73 (875) UBS AG London Banch Sell 7,73 (413) UBS AG London Banch Sell 7,73 (423) UBS AG London Banch Sell 7,73 (423) UBS AG London Banch Sell	AG London Broach	170	707 0	(000)	Clouday
UBS AG London Branch Sell 1,246 (14) UBS AG London Branch Sell 1,246 (16) UBS AG London Branch Sell 9 (1) UBS AG London Branch Sell 855 (665) UBS AG London Branch Sell 214 (24) UBS AG London Branch Sell 249 (73) UBS AG London Branch Sell 249 (73) UBS AG London Branch Sell 2,075 (28) UBS AG London Branch Sell 1,787 (220) UBS AG London Branch Sell 1,977 (271) UBS AG London Branch Sell 1,977 (271) UBS AG London Branch Sell 1,977 (271) UBS AG London Branch Sell 1,977 (443) UBS AG London Branch Sell 1,977 (443) UBS AG London Branch Sell 1,978 (443) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell<	OCCUPATION OF THE PROPERTY OF		3,204	(300)	Ordinary
UBS AG London Branch Sell 1,435 (161) UBS AG London Branch Sell 5,925 (165) UBS AG London Branch Sell 7,128 (800) UBS AG London Branch Sell 8,507 (84) UBS AG London Branch Sell 2,407 (24) UBS AG London Branch Sell 2,075 (233) UBS AG London Branch Sell 2,075 (233) UBS AG London Branch Sell 2,075 (233) UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 5,109 (37) UBS AG London Branch Sell 1,031 (43) UBS AG London Branch Sell 2,442 (37) UBS AG London Branch	AG London Branch	Sell	1,246	(140)	Ordinary
UBS AG London Branch Sell 5,925 (665) UBS AG London Branch Sell 7,128 (800) UBS AG London Branch Sell 7,128 (800) UBS AG London Branch Sell 214 (24) UBS AG London Branch Sell 2,075 (28) UBS AG London Branch Sell 7,001 (81) UBS AG London Branch Sell 7,075 (23) UBS AG London Branch Sell 7,077 (200) UBS AG London Branch Sell 7,077 (27) UBS AG London Branch Sell 5,109 (57) UBS AG London Branch Sell 5,109 (57) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (43) UBS AG London Branch Sell 3,943 (43) UBS AG London Branch Sell 3,943 (47) UBS AG London Branch <	AG London Branch	Seil	1,435	(191)	Ordinary
UBS AG London Branch Sell 1.128 (1) UBS AG London Branch Sell 7.128 (800) UBS AG London Branch Sell 855 (96) UBS AG London Branch Sell 214 (24) UBS AG London Branch Sell 2,075 (233) UBS AG London Branch Sell 2,075 (233) UBS AG London Branch Sell 1,787 (220) UBS AG London Branch Sell 1,977 (221) UBS AG London Branch Sell 1,977 (212) UBS AG London Branch Sell 5,109 (57) UBS AG London Branch Sell 1,931 (116) UBS AG London Branch Sell 1,031 (116) UBS AG London Branch Sell 1,031 (437) UBS AG London Branch Sell 2,44 (278) UBS AG London Branch Sell 1,423 (437) UBS AG London Branch Sell 1,4423 (437) UBS AG London Branch	AG London Branch	Sell	5.975	(999)	Ordinan
UBS AG London Branch Sell 7.128 (800) UBS AG London Branch Sell 214 (84) UBS AG London Branch Sell 249 (28) UBS AG London Branch Sell 2,075 (23) UBS AG London Branch Sell 720 (81) UBS AG London Branch Sell 720 (81) UBS AG London Branch Sell 5,109 (57) UBS AG London Branch Sell 5,109 (57) UBS AG London Branch Sell 1,089 (6) UBS AG London Branch Sell 5,109 (67) UBS AG London Branch Sell 1,068 (1,10) UBS AG London Branch Sell 2,474 (7,22) UBS AG London Branch Sell 3,943 (43) UBS AG London Branch Sell<	AG London Branch	Coll	C	(2)	
UBS AG London Branch Sell 7.128 (800) UBS AG London Branch Sell 2.075 (24) UBS AG London Branch Sell 2.075 (233) UBS AG London Branch Sell 2.075 (233) UBS AG London Branch Sell 2.075 (233) UBS AG London Branch Sell 2.077 (220) UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 5,109 (57) UBS AG London Branch Sell 5,397 (50) UBS AG London Branch Sell 1,031 (116) UBS AG London Branch Sell 2,442 (54) UBS AG London Branch Sell 2,442 (6) UBS AG London Branch Sell 2,423 (437) UBS AG London Branch Sell 2,442 (37) UBS AG London Branch	A 1	nac a	6	(1)	Cidinary
UBS AG London Branch Sell 855 (96) UBS AG London Branch Sell 214 (24) UBS AG London Branch Sell 7,601 (854) UBS AG London Branch Sell 2,075 (233) UBS AG London Branch Sell 2,075 (233) UBS AG London Branch Sell 1,787 (200) UBS AG London Branch Sell 720 (81) UBS AG London Branch Sell 720 (81) UBS AG London Branch Sell 5,541 (266) UBS AG London Branch Sell 1,037 (115) UBS AG London Branch Sell 8,597 (967) UBS AG London Branch Sell 1,068 (437) UBS AG London Branch Sell 1,068 (437) UBS AG London Branch Sell 2,443 (278) UBS AG London Branch Sell 2,443 (278) UBS AG London Branch Sell 2,443 (278) UBS AG London Branch <	AG London Branch	sell	7,128	(800)	Ordinary
UBS AG London Branch Sell 7.49 (24) UBS AG London Branch Sell 7.601 (824) UBS AG London Branch Sell 2.49 (288) UBS AG London Branch Sell 1.781 (200) UBS AG London Branch Sell 1.977 (222) UBS AG London Branch Sell 2.541 (286) UBS AG London Branch Sell 2.541 (286) UBS AG London Branch Sell 2.5109 (575) UBS AG London Branch Sell 1.088 (577) UBS AG London Branch Sell 1.088 (120) UBS AG London Branch Sell 5.39 (687) UBS AG London Branch Sell 5.39 (120) UBS AG London Branch Sell 5.39 (2.28) UBS AG London Branch Sell 4.43 (4.43) UBS AG London Branch Sell 2.474 (2.28) UBS AG London Branch Sell 2.474 (2.28) UBS AG London Branch	AG London Branch	Sell	855	(96)	Ordinary
UBS AG London Branch Sell 7,601 (854) UBS AG London Branch Sell 2,05 (28) UBS AG London Branch Sell 1,781 (200) UBS AG London Branch Sell 1,787 (200) UBS AG London Branch Sell 1,977 (220) UBS AG London Branch Sell 2,741 (280) UBS AG London Branch Sell 5,109 (87) UBS AG London Branch Sell 5,103 (116) UBS AG London Branch Sell 4,87 (43) UBS AG London Branch Sell 4,87 (43) UBS AG London Branch Sell 3,93 (43) UBS AG London Branch Sell 3,943 (43) UBS AG London Branch Sell 2,893 (43) UBS AG London Branch Sell 2,442 (22) UBS AG London Branch Sell 2,443 (32) UBS AG London Branch Sell 2,443 (43) UBS AG London Branch <th< td=""><td>AG London Branch</td><td>Sell</td><td>214</td><td>(24)</td><td>Ordinary</td></th<>	AG London Branch	Sell	214	(24)	Ordinary
UBS AG London Branch Sell 7,00 UBS AG London Branch Sell 1,00 UBS AG London Branch Sell 1,00 UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 2,541 (286) UBS AG London Branch Sell 2,541 (286) UBS AG London Branch Sell 1,03 (110) UBS AG London Branch Sell 1,08 (110) UBS AG London Branch Sell 1,08 (120) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 2,474 (2,78) UBS AG London Branch Sell 2,474 (2,78) UBS AG London Branch Sell 2,474 (2,78)	AG London Branch	Coll	7 601	(00.4)	0.00
UBS AG London Branch Sell 2.075 (238) UBS AG London Branch Sell 2.075 (237) UBS AG London Branch Sell 1,781 (200) UBS AG London Branch Sell 1,781 (201) UBS AG London Branch Sell 2,541 (285) UBS AG London Branch Sell 5,103 (115) UBS AG London Branch Sell 1,031 (115) UBS AG London Branch Sell 8,597 (867) UBS AG London Branch Sell 1,688 (120) UBS AG London Branch Sell 1,688 (120) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 2,744 (278) UBS AG London Branch Sell 2,744 (87) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 3,552 (370) UBS AG London Branch <td>A COLOR DISTOR</td> <td>201</td> <td>100'/</td> <td>(400)</td> <td>Ordinary</td>	A COLOR DISTOR	201	100'/	(400)	Ordinary
UBS AG London Branch Sell 1,735 (233) UBS AG London Branch Sell 1,587 (202) UBS AG London Branch Sell 7,077 (871) UBS AG London Branch Sell 7,271 (871) UBS AG London Branch Sell 2,541 (286) UBS AG London Branch Sell 1,031 (116) UBS AG London Branch Sell 1,031 (116) UBS AG London Branch Sell 1,032 (6) UBS AG London Branch Sell 3,943 (4,43) UBS AG London Branch Sell 3,943 (4,43) UBS AG London Branch Sell 3,943 (4,43) UBS AG London Branch Sell 3,943 (4,13) UBS AG London Branch Sell 2,474 (2,78) UBS AG London Branch Sell 2,474 (2,78) UBS AG London Branch Sell 7,449 (8) UBS AG London Branch Sell 7,449 (8) UBS AG London Branch <td>AG LONGON Branch</td> <td>Seil</td> <td>249</td> <td>(78)</td> <td>Ordinary</td>	AG LONGON Branch	Seil	249	(78)	Ordinary
UBS AG London Branch Sell 5.085 (57) UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 2,541 (158) UBS AG London Branch Sell 5,109 (375) UBS AG London Branch Sell 4,597 (497) UBS AG London Branch Sell 4,597 (497) UBS AG London Branch Sell 1,088 (413) UBS AG London Branch Sell 1,088 (437) UBS AG London Branch Sell 3,443 (443) UBS AG London Branch Sell 3,43 (443) UBS AG London Branch Sell 2,74 (278) UBS AG London Branch Sell 3,43 (443) UBS AG London Branch Sell 7,43 (437) UBS AG London Branch Sell 7,43 (437) UBS AG London Branch Sell 7,43 (437) UBS AG London Branch	AG London Branch	Sell	2,075	(233)	Ordinary
UBS AG London Branch Sell 5,065 (571) UBS AG London Branch Sell 1,977 (222) UBS AG London Branch Sell 2,541 (286) UBS AG London Branch Sell 5,109 (575) UBS AG London Branch Sell 8,597 (467) UBS AG London Branch Sell 8,597 (467) UBS AG London Branch Sell 1,688 (100) UBS AG London Branch Sell 53 (6) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (43) UBS AG London Branch Sell 3,943 (43) UBS AG London Branch Sell 2,474 (176) UBS AG London Branch Sell 2,474 (176) UBS AG London Branch Sell 2,552 (30) UBS AG London Branch Sell 2,568 (32) UBS AG London Branch	AG London Branch	Sell	1,781	(200)	Ordinary
UBS AG Landon Branch Sell 1,577 (222) UBS AG Landon Branch Sell 1,577 (222) UBS AG Landon Branch Sell 2,541 (386) UBS AG Landon Branch Sell 5,109 (575) UBS AG Landon Branch Sell 4,597 (16) UBS AG Landon Branch Sell 1,531 (116) UBS AG Landon Branch Sell 1,58 (22) UBS AG Landon Branch Sell 3,943 (4,43) UBS AG Landon Branch Sell 2,893 (437) UBS AG Landon Branch Sell 2,474 (278) UBS AG Landon Branch Sell 1,423 (437) UBS AG Landon Branch Sell 1,442 (437) UBS AG Landon Branch Sell 1,442 (370) UBS AG Landon Branch Sell 1,442 (437) UBS AG Landon Branch Sell 2,474 (278) UBS AG Landon Branch Sell 3,523 (430) UBS AG Landon Branch	AG London Branch	llay	7807	(571)	Ordinary
UBS AG London Branch Sell 1,977 (22) UBS AG London Branch Sell 2,541 (286) UBS AG London Branch Sell 5,109 (575) UBS AG London Branch Sell 8,597 (967) UBS AG London Branch Sell 8,597 (967) UBS AG London Branch Sell 8,597 (967) UBS AG London Branch Sell 5,3 (6) UBS AG London Branch Sell 5,3 (6) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (437) UBS AG London Branch Sell 2,474 (7,78) UBS AG London Branch Sell 7,449 (7) UBS AG London Branch Sell 7,449 (7) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 7,649 (7) UBS AG London Branch <td< td=""><td>AC 10 10 10 10 10 10 10 10 10 10 10 10 10</td><td></td><td>COO'C</td><td>(1/0)</td><td>Ciginally</td></td<>	AC 10 10 10 10 10 10 10 10 10 10 10 10 10		COO'C	(1/0)	Ciginally
UBS AG London Branch Sell 2,541 (359) UBS AG London Branch Sell 2,541 (375) UBS AG London Branch Sell 5,109 (575) UBS AG London Branch Sell 8,597 (44) UBS AG London Branch Sell 8,597 (45) UBS AG London Branch Sell 196 (22) UBS AG London Branch Sell 1,088 (1,20) UBS AG London Branch Sell 3,943 (44) UBS AG London Branch Sell 2,893 (497) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 2,474 (1,776) UBS AG London Branch Sell 2,474 (370) UBS AG London Branch Sell 2,474 (370) UBS AG London Branch Sell 2,473 (497) UBS AG London Branch	AG London Branch	Xe.ii	//6'L	(222)	Ordinary
UBS AG London Branch Sell 2,541 (286) UBS AG London Branch Sell 5,109 (573) UBS AG London Branch Sell 8,597 (967) UBS AG London Branch Sell 8,597 (967) UBS AG London Branch Sell 1,08 (22) UBS AG London Branch Sell 53 (6) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 2,474 (176) UBS AG London Branch Sell 2,352 (497) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch	AG London Branch	Sell	720	(81)	Ordinary
UBS AG London Branch Sell 5,109 (575) UBS AG London Branch Sell 8,591 (416) UBS AG London Branch Sell 8,591 (427) UBS AG London Branch Sell 48,191 (54) UBS AG London Branch Sell 1,068 (120) UBS AG London Branch Sell 1,068 (120) UBS AG London Branch Sell 2,893 (432) UBS AG London Branch Sell 2,893 (432) UBS AG London Branch Sell 2,474 (2,78) UBS AG London Branch Sell 1,423 (437) UBS AG London Branch Sell 1,478 (1,176) UBS AG London Branch Sell 2,474 (2,78) UBS AG London Branch Sell 2,352 (340) UBS AG London Branc	AG London Branch	Seil	2.541	(786)	Ordinary
UBS AG London Branch Sell 1,031 (15) UBS AG London Branch Sell 48.1 (48.7) UBS AG London Branch Sell 48.1 (54) UBS AG London Branch Sell 1.068 (1.20) UBS AG London Branch Sell 3.943 (4.43) UBS AG London Branch Sell 3.943 (4.23) UBS AG London Branch Sell 2.793 (3.75) UBS AG London Branch Sell 2.474 (2.78) UBS AG London Branch Sell 2.474 (2.78) UBS AG London Branch Sell 2.474 (2.78) UBS AG London Branch Sell 2.474 (3.78) UBS AG London Branch Sell 2.653 (49) UBS AG London Branch Sell 3.297 (370) UBS AG London Branc	AG London Branch	100	1,77	1200	1000
UBS AG London Branch Sell (116) UBS AG London Branch Sell 481 (54) UBS AG London Branch Sell 481 (54) UBS AG London Branch Sell 53 (66) UBS AG London Branch Sell 53 (6) UBS AG London Branch Sell 3,943 (120) UBS AG London Branch Sell 2,893 (423) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 1,6 (477) UBS AG London Branch Sell 1,749 (1,176) UBS AG London Branch Sell 1,0,478 (1,176) UBS AG London Branch Sell 2,352 (24) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 3,297 (310) UBS AG London Branch Sell 2,352 (44) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell <td< td=""><td>AC COLOCI DISTILLI</td><td>360</td><td>50,10</td><td>(5/5)</td><td>Ordinary</td></td<>	AC COLOCI DISTILLI	360	50,10	(5/5)	Ordinary
UBS AG London Branch Sell 48.597 (967) UBS AG London Branch Sell 48.1 (34) UBS AG London Branch Sell 1.068 (120) UBS AG London Branch Sell 5.3 (6) UBS AG London Branch Sell 3.943 (4.13) UBS AG London Branch Sell 2.893 (4.35) UBS AG London Branch Sell 2.893 (4.37) UBS AG London Branch Sell 2.474 (2.78) UBS AG London Branch Sell 2.474 (2.78) UBS AG London Branch Sell 7.449 (8.50) UBS AG London Branch Sell 7.449 (8.50) UBS AG London Branch Sell 7.649 (8.51) UBS AG London Branch Sell 7.638 (4.10) UBS AG London Branch Sell 7.638 (8.64) UBS AG London Branch Sell 7.638 (8.64) UBS AG London Branch Sell 7.638 (8.64) UBS AG London Br		Sell	1,031	(116)	Ordinary
UBS AG London Branch Sell 481 (54) UBS AG London Branch Sell 196 (122) UBS AG London Branch Sell 1,068 (120) UBS AG London Branch Sell 3,943 (432) UBS AG London Branch Sell 2,893 (457) UBS AG London Branch Sell 2,893 (432) UBS AG London Branch Sell 1,176 (437) UBS AG London Branch Sell 1,176 (1,176) UBS AG London Branch Sell 1,2474 (1,176) UBS AG London Branch Sell 1,2474 (1,176) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch		Sell	8,597	(296)	Ordinary
UBS AG London Branch Sell 196 (22) UBS AG London Branch Sell 1,088 (1,00) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 2,893 (432) UBS AG London Branch Sell 2,893 (432) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 1,176 (497) UBS AG London Branch Sell 1,176 (1,176) UBS AG London Branch Sell 1,443 (1,176) UBS AG London Branch Sell 1,443 (1,176) UBS AG London Branch Sell 2,352 (350) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 7,638 (843) UBS AG London Branch Sell 7,638 (843) UBS AG London Branch Sell 7,638 (843) UBS AG London Branc		Sell	481	(54)	Ordinary
UBS AG Landon Branch Sell 1,558 (5) UBS AG Landon Branch Sell 1,068 (120) UBS AG Landon Branch Sell 3,943 (443) UBS AG Landon Branch Sell 3,943 (443) UBS AG Landon Branch Sell 4,233 (325) UBS AG London Branch Sell 2,474 (2,78) UBS AG London Branch Sell 1,0479 (1,156) UBS AG London Branch Sell 2,474 (1,156) UBS AG London Branch Sell 2,474 (1,156) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 2,698 (864) UBS AG London Branch Sell 2,698 (864) UBS AG London Branch Sell 1,710 (1,22) UBS AG London Br		Coll	196	((()	Cadinan
UBS AG London Branch Sell 153 (b) UBS AG London Branch Sell 1,058 (120) UBS AG London Branch Sell 2,893 (433) UBS AG London Branch Sell 2,893 (325) UBS AG London Branch Sell 4,423 (497) UBS AG London Branch Sell 1,176 UBS AG London Branch Sell 1,474 (278) UBS AG London Branch Sell 1,478 (1,176) UBS AG London Branch Sell 2,352 (497) UBS AG London Branch Sell 2,352 (497) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 7,598 (844) UBS AG London Branch Sell 7,598 (842) UBS AG London Branch Sell 7,598 (842) UBS AG London Branch Sell 7,598 (842) UBS AG London Branch Sell		100	061	(22)	Ciginaly
UBS AG London Branch Sell 1,088 (120) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 4,423 (437) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 1,449 (178) UBS AG London Branch Sell 7,449 (176) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 2,876 (323) UBS AG London Branch Sell 1,211 (1,35) UBS AG London Branch Sell 1,211 (1,32) UBS AG London Branch Sell 1,211 (1,32) UBS AG London Branch	Ad Colode Blaffel	ner.	53	(0)	Ordinary
UBS AG London Branch Sell 3,943 (6) UBS AG London Branch Sell 2,893 (437) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 1,474 (278) UBS AG London Branch Sell 1,474 (278) UBS AG London Branch Sell 1,479 (1,176) UBS AG London Branch Sell 2,352 (349) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 3,327 (41) UBS AG London Branch Sell 3,257 (430) UBS AG London Branch Sell 3,257 (430) UBS AG London Branch Sell 7,589 (864) UBS AG London Branch Sell 3,871 (430) UBS AG London Branch Sell 7,589 (864) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 1,211 (198) UBS AG London Branch <td>AG London Branch</td> <td>Sell</td> <td>1,068</td> <td>(120)</td> <td>Ordinary</td>	AG London Branch	Sell	1,068	(120)	Ordinary
UBS AG London Branch Sell 3,943 (443) UBS AG London Branch Sell 4,283 (325) UBS AG London Branch Sell 4,474 (278) UBS AG London Branch Sell 1,8 (1,176) UBS AG London Branch Sell 1,449 (1,176) UBS AG London Branch Sell 7,449 (1,176) UBS AG London Branch Sell 2,152 (350) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 7,638 (643) UBS AG London Branch Sell 7,638 (643) UBS AG London Branch Sell 1,211 (135) UBS AG London Branch Sell 1,211 (135) UBS AG London Branch Sell 1,211 (135) UBS AG London Branc	AG London Branch	Sell	23	(9)	Ordinary
UBS AG London Branch Sell 2,893 (325) UBS AG London Branch Sell 4,423 (497) UBS AG London Branch Sell 1,478 (1,716) UBS AG London Branch Sell 10,478 (1,176) UBS AG London Branch Sell 2,449 (885) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 3,297 (1) UBS AG London Branch Sell 3,297 (1) UBS AG London Branch Sell 3,297 (430) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 2,876 (841) UBS AG London Branch Sell 1,710 (135) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch </td <td>AG London Branch</td> <td>Sell</td> <td>3 943</td> <td>(577)</td> <td>Ordinan</td>	AG London Branch	Sell	3 943	(577)	Ordinan
UBS AG London Branch Sell 4,422 (427) UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 1,8 (7,176) UBS AG London Branch Sell 7,49 (856) UBS AG London Branch Sell 7,49 (856) UBS AG London Branch Sell 2,352 (370) UBS AG London Branch Sell 7,292 (81) UBS AG London Branch Sell 7,638 (1) UBS AG London Branch Sell 7,638 (84) UBS AG London Branch Sell 7,638 (84) UBS AG London Branch Sell 7,638 (84) UBS AG London Branch Sell 7,638 (130) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch	AG London Branch	3	2 803	(305)	Ording
UBS AG London Branch Sell 4,423 (497) UBS AG London Branch Sell 1,3478 (1,176) UBS AG London Branch Sell 1,0478 (1,176) UBS AG London Branch Sell 7,449 (8,65) UBS AG London Branch Sell 2,352 (3,50) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 3,297 (430) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 2,876 (84) UBS AG London Branch Sell 2,876 (430) UBS AG London Branch Sell 1,211 (130) UBS AG London Br	Ac London Branch	l so	2007	(TOT)	(in in in
UBS AG London Branch Sell 2,474 (278) UBS AG London Branch Sell 18 (27) UBS AG London Branch Sell 7,449 (835) UBS AG London Branch Sell 2,352 (749) UBS AG London Branch Sell 2,352 (749) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 722 (81) UBS AG London Branch Sell 7,598 (864) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 7,598 (864) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch	0.0000000000000000000000000000000000000	201	4,442	(48/)	Crainary
UBS AG London Branch Sell 18 (2) UBS AG London Branch Sell 1,478 (1,176) UBS AG London Branch Sell 2,352 (364) UBS AG London Branch Sell 2,357 (370) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 722 (81) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 2,876 (323) UBS AG London Branch Sell 1,711 (136) UBS AG London Branch Sell 2,876 (323) UBS AG London Branch Sell 1,711 (136) UBS AG London Branch Sell 1,710 (135) UBS AG London Branch	AG London Branch	Sell	2,474	(278)	Ordinary
UBS AG London Branch Sell 10,478 (1,176) UBS AG London Branch Sell 2,449 (836) UBS AG London Branch Sell 2,357 (370) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 722 (81) UBS AG London Branch Sell 722 (81) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 2,876 (824) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,710 (192)	AG London Branch	Sell	18	(2)	Ordinary
UBS AG London Branch Sell 7,449 (17.49) UBS AG London Branch Sell 2,352 (264) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 722 (81) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 7,698 (864) UBS AG London Branch Sell 7,698 (864) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch <td>AG London Branch</td> <td>la S</td> <td>10.478</td> <td>(1 176)</td> <td>Ordinan</td>	AG London Branch	la S	10.478	(1 176)	Ordinan
UBS AG London Branch Sell 2.352 (264) UBS AG London Branch Sell 2.352 (270) UBS AG London Branch Sell 3.297 (370) UBS AG London Branch Sell 772 (81) UBS AG London Branch Sell 722 (81) UBS AG London Branch Sell 7.698 (864) UBS AG London Branch Sell 7.297 (871) UBS AG London Branch Sell 1.211 (135) UBS AG London Branch Sell 1.711 (136) UBS AG London Branch Sell 1.710 (192) Manch Ag London Branch Sell 1.710 (192)	1000000		0/1/01	(0/1/1)	Collialy
UBS AG London Branch Sell 2.352 (264) UBS AG London Branch Sell 3,297 (370) UBS AG London Branch Sell 722 (81) UBS AG London Branch Sell 3,831 (40) UBS AG London Branch Sell 3,831 (40) UBS AG London Branch Sell 7,698 (864) UBS AG London Branch Sell 7,698 (864) UBS AG London Branch Sell 1,710 (192)	Ab London Branch	Seli	7,449	(836)	Ordinary
UBS AG London Branch Sell 3.297 (370) UBS AG London Branch Sell 772 (81) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 2,876 (842) UBS AG London Branch Sell 2,876 (842) UBS AG London Branch Sell 2,876 (323) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,210 (192)	AG London Branch	Self	2,352	(564)	Ordinary
UBS AG London Branch Sell 722 (1) UBS AG London Branch Sell 722 (81) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 7,638 (864) UBS AG London Branch Sell 7,876 (323) UBS AG London Branch Sell 1,211 (135) UBS AG London Branch Sell 1,211 (135) UBS AG London Branch Sell 1,710 (192)	AG London Branch	Sell	3,297	(370)	Ordinary
UBS AG London Branch Sell 722 (4) UBS AG London Branch Sell 3,831 (430) UBS AG London Branch Sell 3,875 (84) UBS AG London Branch Sell 2,876 (32) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 5,236 (588) UBS AG London Branch Sell 5,236 (588) UBS AG London Branch Sell 5,236 (588)	AG London Branch	Call	a	(4)	Ordina
USS AGL London Branch Sell 7.22 (81) USS AGL London Branch Sell 7,598 (864) UBS AGL London Branch Sell 2,876 (852) UBS AGL London Branch Sell 1,211 (135) UBS AGL London Branch Sell 1,710 (192)	Ac london Dranch	117	6.65		o durially
UBS AG London Branch Sell 3,831 (430) USS AG London Branch Sell 7,698 (644) UBS AG London Branch Sell 2,876 (323) UBS AG London Branch Sell 1,211 (130) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 5,236 (589) HIR AG London Branch Sell 5,236 (589) HIR AG London Branch Sell 5,236 (589)	AG LUIGOI BIAIRCI	nei(77/	(18)	Ordinary
USS AG London Branch Sell 7.688 (864) UBS AG London Branch Sell 2,876 (323) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 5,236 (588) UBS AG London Branch Sell 5,236 (588)	AG London Branch	Sell	3,831	(430)	Ordinary
UBS AG London Branch Sell 2,876 (323) UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 5,236 (589) IRS AG London Branch Sell 5,236 (589)	AG London Branch	Set	7.698	(864)	Ordinary
Control Control Reserve Cell 4,507 to 135 UBS AG London Branch Sell 1,211 (135) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 5,236 (588) UBS AG London Branch Sell 5,236 (389)	AG London Branch	ll o	370 ((606)	
UBS AG London Branch Sell 1,211 (136) UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 5,236 (589) IIRS AG I morkin Branch Sell 5,236 (589)	C C C C C C C C C C C C C C C C C C C	2011	2,070	(676)	Cignary
UBS AG London Branch Sell 1,710 (192) UBS AG London Branch Sell 5,236 (589) I IRS AG London Branch Sell 5,236 (580)	AG London Branch	Sell	1,211	(136)	Ordinary
UBS AG London Branch Sell 5,236 (588)	AG London Branch	Zell Sell	1,710	(192)	Ordinary
I IRC 46 London Branch Cell 2 6727 (2001)	AG London Branch	Sell	5.236	(5,88)	Ordinary
	A Charles Described		5000	(200)	1000
מסכן איים ביים מיים ביים ביים ביים ביים ביים ב	Als London Branch	Sei	7/9/7	(300)	Ordinary
07-May-14 LIBS				Self S	Self 640 Self 642 Self 6,429 Self 6,229 Self 7,246 Self 7,246 Self 7,246 Self 7,246 Self 7,246 Self 7,246 Self 7,241 Self 7,241 Self 7,241 Self 7,249 Self 7

T	UBS Securities Australia Ltd	Sell	13,455	(1,500)	Ordinary
T	UBS Securities Australia Ltd	Sell	1,041	(116)	Ordinary
T	JBS Securities Australia Ltd	lles.	2,090	(233)	Ordinary
7	UBS Securities Australia Ltd	Sell	6	E.	Ordinary
06-May-14	UBS Securities Australia Ltd	Sell	20,057	(2,236)	Ordinary
7	JBS Securities Australia Ltd	Selt	26,327	(2,935)	Ordinary
Ť	UBS Securities Australia Ltd	Seli	3,687	(411)	Ordinary
┪	JBS Securities Australia Ltd	Sel	179	(20)	Ordinary
06-May-14	UBS Securities Australia Ltd	Sell	6	(1)	Ordinary
\neg	JBS Securities Australia Ltd	Sell	1,121	(125)	Ordinary
06-May-14	JBS Securities Australia Ltd	Sell	18	(2)	Ordinary
06-May-14	JBS Securities Australia Ltd	Sell	2.269	(253)	Ordinary
06-May-14	UBS Securities Australia Ltd	Sell	727	(81)	Ordinary
1	IBS Securities Australia Ltd	lo)	915	(201)	Ordinan
06-May-14	IDC Cocurition Americal and	123	2 404 C	(201)	Cidinary
	Do Securities Australia Lid	nac o	2,434	(0/7)	Ordinary
Ť	JBS Securities Australia Ltd	Sell	24,040	(089/7)	Ordinary
Ť	JBS Securities Australia Ltd	<u> </u>	4,422	(493)	Ordinary
7	JBS Securities Australia Ltd	<u></u>	6,458	(720)	Ordinary
	UBS Securities Australia Ltd	Sell	2,512	(280)	Ordinary
	JBS Securities Australia Ltd	Setl	1,579	(176)	Ordinary
06-May-14 ≀	UBS Securities Australia Ltd	Sett	3,202	(357)	Ordinary
	JBS Securities Australia Ltd	Sei	2,565	(286)	Ordinary
l	UBS Securities Australia Ltd	Sell	5.176	(222)	Ordinary
r	UBS Securities Australia Ltd	Sell	1665	(582)	Ordinary
06-May-14	UBS Securities Australia Ltd	9	10.495	(1170)	Ordinan
T	IRS Sequenties Australia 14d		22.25	(0/1/1)	Ordinary
T	IRA Cocurition Americalia 142	100	4 613	(503)	Ordinary
06 May 14	IDC Countries Asserting Little	liec liec	216,4	(505)	Ordinary
Ť	DD Securities Australia Ltd	Oeti	757,77	(2,473)	Ordinary
1	UBS Securities Australia Ltd	Seil	53,425	(5,956)	Ordinary
1	UBS Securities Australia Ltd	Sell	13,527	(1,508)	Ordinary
	UBS Securities Australia Ltd	Sell	23,977	(2,673)	Ordinary
-May-14	JBS Securities Australia Ltd	Sell	57,614	(6,423)	Ordinary
ı	UBS Securities Australia Ltd	Sell	2,969	(331)	Ordinary
1	IRS Securities Australia Ltd	Sell	60 180	(6.710)	Ordinan
1	IBS Securities Australia I to	3	1 032	(115)	Ordinary
Ť	IBC Consistion Australia 14d	l coll	1631	(5.1)	Cidirialy
+	DO Securities Australia Ltd	100	7/9'1	(/0))	Ordinary
1	Obsideration Australia Lto	ne.	6/-	(NZ)	Ordinary
Ub-May-14	UBS Securities Australia Ltd	Seil	30,462	(3,396)	Ordinary
┪	JBS Securities Australia Ltd	Sell	5,292	(280)	Ordinary
Ť	JBS Securities Australia Ltd	Sell	4,485	(200)	Ordinary
٦	JBS Securities Australia Ltd	Sell	7,849	(875)	Ordinary
_	UBS Securities Australia Ltd	Sell	16,899	(1,884)	Ordinary
Ť	UBS Securities Australia Ltd	Self	1,094	(122)	Ordinary
Ī	UBS Securities Australia Ltd	Sell	2,646	(562)	Ordinary
06-May-14	BS Securities Australia I to	9	14 836	(1,654)	Ordinary
Ť	MC Consisting Australia Ltd	120	C14, CC	(+00,1)	Clumary
Ť	203 Seculities Australia Liu	196	75,457	(100(5)	Orginary
Ť	JBS securities Australia Ltd	Sell	46,554	(5,190)	Ordinary
	JBS Securities Australia Ltd	Sell	6,539	(729)	Ordinary
7	JBS Securities Australia Ltd	Seti	4,646	(518)	Ordinary
_	JBS Securities Australia Ltd	Sell	1,023	(114)	Ordinary
۲	JBS Securities Australia Ltd	Sell	13,069	(1,457)	Ordinary
T	JBS Securities Australia Ltd	laS.	2 063	(087)	Ordinary
Ť	IBC Sequities Australia 14d	lo S	2 706	(000)	Ordinary
Ť	IDC Commission Australia Lea	11-0	08/1/	(000)	Gullary
Ť	Jos securities Australia Ltd) sell	39,549	(4,409)	Ordinary
_	J85 Securities Australia Ltd	Sell	58,305	(6,500)	Ordinary
7	JBS Securities Australia Ltd	Sell	493	(25)	Ordinary
╗	UBS Securities Australia Ltd	Sell	4,485	(200)	Ordinary
06-May-14	UBS Securities Australia Ltd	Sell	4,808	(989)	Ordinary
06-May-14	UBS Securities Australia Ltd	Sell	736	(82)	Ordinary
Г	IBS Securities Australia Ltd	Sell	897	(100)	Ordinary
06-May-14	UBS Securities Australia Ltd	le ₂	51 003	(5 686)	Ordinary
T	IRS Sequeties Australia I to	- -	9399	(000,0)	Ordinary.
T	Por countries Australia Liu	= -	959'9	(/42)	Ordinary
Ť	Jos securities Australia Ltd	Sell	1,749	(cs.)	Ordinary
7	UBS Securities Australia Ltd	Sell	17.438	(1 000)	C. C.
06 May 14				1,440,11	Ordinary
	UBS Securities Australia Ltd	Ş	3,256	(363)	Ordinary

Box Box	Name 14	I IDC Committee America 144				Ordinary
USS Securities Autorials Ltd Self Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Autorials Ltd Self Securities Au	-May-14	LIBS Securities Australia Ind	Sell	190,66	0000	1
185 Securities Australia Ltd 181	-Mav-14	UBS Securities Australia 11d	× 100	336 008	36 090	Ordinary
Securites Australia Ltd Self 1,000 1,0	-May-14	LIBS Securities Australia 14d	(a)	216 524	1000 607	Cidirlary
1982 Securines Autoritial Ltd 540 10,000 1,000	-May-14	I IBC Contribute Australia 14d	100	420,017	(25,009)	Ordinary
1985 Securities Australia Ltd Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self Self 1,3,000 1985 Securities Australia Ltd Self 1,0,000 0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd Self 1,0,000 1,0,000 1985 Securities Australia Ltd	11	I De countries Australia Ltd)na	412,284	45,300	Ordinary
Executive Autorials Ltd Self Secutive Autorials Ltd Se	May 14	LISC Societies Australia tad	llac lac	100,124	(12,000)	Ordinary
Desi Securities Australia Ltd Buy 552,103 58,410 Lobis Securities Australia Ltd Buy 37,103 58,410 Lobis Securities Australia Ltd Buy 3,731 410 Lobis Securities Australia Ltd Buy 3,731 410 Lobis Securities Australia Ltd Buy 3,731 410 Lobis Securities Australia Ltd Sell 1,010,299 (1,138) Lobis Securities Australia Ltd Sell 1,029 (1,138) Lobis Securities Australia Ltd Sell 2,748 (660) Lobis Securities Australia Ltd Sell 7,817 (663) Lobis Securities Australia Ltd Sell 7,817 (663) Lobis Securities Australia Ltd Sell 7,817 (663) Lobis Securities Australia Ltd Sell 7,817 (673) Lobis Securities Australia Ltd Sell 7,817 (673) Lobis Securities Australia Ltd Sell 7,827 (673) Lobis Securities Australia Ltd Sell 7,827 (673) <	May-14	LIBC Counties Australia 14d	pay.	41,930	4,050	Ordinary
Cost Securities Australia Ltd Sayl 1,910 C\$10 DIS Gobbil Assis Namagement Sayl 1,733 410 105 DIS Schull Randagement Bay 3,733 410 105 DIS Schuller Australia Ltd Sele 82,206 9,820 410 DIS Schuller Australia Ltd Sele 3,733 410 10,239 (1,118) DIS Schuller Australia Ltd Sele 3,733 410 10,239 (1,118) DIS Schuller Australia Ltd Sele 3,734 410 10,239 (1,138) DIS Schriffer Australia Ltd Sele 1,029 (1,138) (1,138) (1,138) DIS Schriffer Australia Ltd Sele 1,029 (1,138) (1,249) (1,138) DIS Schriffer Australia Ltd Sele 1,029 (1,128) (1,229) (1,128) DIS Schriffer Australia Ltd Sele 1,026 (1,249) (1,249) (1,249) DIS Schriffer Australia Ltd Sele 1,029 (1,128) (1,249) (1,249)	Mon 14	LIDS Contribute Australia Lida	liac li	905,51	(000,1)	Ordinary
USS Securities Australia Ltd Sept 1,373 1,000	May 14	LIDE Clobal Acces Management	bauy Sell	532,103	58,410	Ordinary
USS Securities Australia Ltd Self	Many 14	COS CIODAI Asset Maliagerifetti	= 2	018,1	(210)	Ordinary
UBS Securities Australia Ltd Bildy 8,746 6,653 UBS Securities Australia Ltd Buy 3,731 410 UBS Securities Australia Ltd Sell 10,299 (1,133) UBS Securities Australia Ltd Sell 10,299 (1,133) UBS Securities Australia Ltd Sell 7,817 (860) UBS Securities Australia Ltd Sell 1,7,89 (1,239) UBS Securities Australia Ltd Sell 1,0,50 (1,239) UBS Securities Australia Ltd Sell 1,0,50 (1,239) UBS Securities Australia Ltd Sell 1,0,50 (1,230) UBS Securities Australia Ltd Sell 1,0,50 (1,230) UBS Securities Australia Ltd Sell 1,0,60 (1,230) UBS Se	P-May-14	UBS Wealth Management Australia Ltd	Buy	3,723	410	Ordinary
UBS Securities Australia Ltd Sell 88,206 (9,539) UBS Securities Australia Ltd Sell 13,731 410 UBS Securities Australia Ltd Sell 10,169 (1,18) UBS Securities Australia Ltd Sell 13,699 (1,18) UBS Securities Australia Ltd Sell 12,699 (1,18) UBS Securities Australia Ltd Sell 7,817 (860) UBS Securities Australia Ltd Sell 7,817 (860) UBS Securities Australia Ltd Sell 7,877 (861) UBS Securities Australia Ltd Sell 1,787 (862) UBS Securities Australia Ltd Sell 1,787 (863) UBS Securities Australia Ltd Sell 1,787 (863) UBS Securities Australia Ltd Sell 1,207 (1,19) UBS Securities Australia Ltd Sell 1,207 (1,19) UBS Securities Australia Ltd Sell 1,207 (1,294) UBS Securities Australia Ltd Sell 1,207 (1,294) UBS Securities	-ivlay-14	UBS Securities Australia Ltd	Buy	/46	82	Ordinary
UBS Securities Australia Ltd Sell 13,731 410 UBS Securities Australia Ltd Sell 10,299 (1,138) UBS Securities Australia Ltd Sell 12,099 (1,138) UBS Securities Australia Ltd Sell 7,817 (860) UBS Securities Australia Ltd Sell 7,877 (860) UBS Securities Australia Ltd Sell 7,787 (860) UBS Securities Australia Ltd Sell 1,782 (7,29) UBS Securities Australia Ltd Sell 1,782 (7,29) UBS Securities Australia Ltd Sell 1,782 (7,29) UBS Securities Australia Ltd Sell 1,782 (7,59) UBS Securities Australia Ltd Sell 1,200 (1,70) UBS Securities Australia Ltd Sell 1,200 (1,70) UBS Securities Au	3-May-14	UBS Securities Australia Ltd	Sel	88,206	(6,693)	Ordinary
UBS Securities Australia Ltd Sell 10,163 (1,181) UBS Securities Australia Ltd Sell 12,099 (1,131) UBS Securities Australia Ltd Sell 3,456 (1,331) UBS Securities Australia Ltd Sell 7,817 (860) UBS Securities Australia Ltd Sell 7,607 (817) UBS Securities Australia Ltd Sell 1,1,562 (1,299) UBS Securities Australia Ltd Sell 1,1,562 (1,249) UBS Securities Australia Ltd Sell 1,1,562 (1,249) UBS Securities Australia Ltd Sell 1,1,562 (1,249) UBS Securities Australia Ltd Sell 1,2,072 (1,509) UBS Securities Australia Ltd Sell 1,2,072 (1,509) UBS Securities Australia Ltd Sell 1,2,072 (1,509)	s-May-14	UBS Securities Australia Ltd	Buy	3,731	410	Ordinary
UBS Securities Australia Ltd Sell 10,299 (1,131) UBS Securities Australia Ltd Sell 3,459 (1,311) UBS Securities Australia Ltd Sell 7,817 (860) UBS Securities Australia Ltd Sell 7,817 (860) UBS Securities Australia Ltd Sell 7,617 (860) UBS Securities Australia Ltd Sell 7,617 (860) UBS Securities Australia Ltd Sell 1,757 (860) UBS Securities Australia Ltd Sell 1,767 (860) UBS Securities Australia Ltd Sell 1,787 (870) UBS Securities Australia Ltd Sell 1,787 (870) UBS Securities Australia Ltd Sell 1,787 (870) UBS Securities Australia Ltd Sell 1,787 (701) UBS Securities Australi	-May-14	UBS Securities Australia Ltd	Sell	10,163	(1,118)	Ordinary
UBS Securities Australia Ltd Self 3,458 (137) UBS Securities Australia Ltd Self 12,099 (1,331) UBS Securities Australia Ltd Self 7,817 (860) UBS Securities Australia Ltd Self 7,817 (860) UBS Securities Australia Ltd Self 7,807 (861) UBS Securities Australia Ltd Self 7,807 (862) UBS Securities Australia Ltd Self 1,752 (2,500) UBS Securities Australia Ltd Self 1,762 (1,249) UBS Securities Australia Ltd Self 1,762 (1,219) UBS Securities Australia Ltd Self 1,620 (1,249) UBS Securities Australia Ltd Self 1,627 (1,219) UBS Securities Australia Ltd Self 1,627 (1,219) UBS Securities Australia Ltd Self 1,627 (1,219) UBS Securities Australia Ltd Self 1,627 (1,721) UBS Securities Australia Ltd Self 1,627 (1,721) UBS Sec	-May-14	UBS Securities Australia Ltd	Sell	10,299	(1,133)	Ordinary
UBS Securities Australia Ltd Sell 13.56 (3.17) UBS Securities Australia Ltd Sell 12.09 (1.31) UBS Securities Australia Ltd Sell 7.817 (865) UBS Securities Australia Ltd Sell 7.408 (863) UBS Securities Australia Ltd Sell 1.757 (863) UBS Securities Australia Ltd Sell 1.787 (863) UBS Securities Australia Ltd Sell 1.787 (863) UBS Securities Australia Ltd Sell 1.787 (863) UBS Securities Australia Ltd Sell 1.0,583 (1.209) UBS Securities Australia Ltd Sell 1.0,683 (1.743) UBS Securities Australia Ltd Sell 1.0,683 (1.743) <t< td=""><td>3-May-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>3,458</td><td>(380)</td><td>Ordinary</td></t<>	3-May-14	UBS Securities Australia Ltd	Sell	3,458	(380)	Ordinary
UBS Securities Australia Ltd Sell 7,877 (850) UBS Securities Australia Ltd Sell 7,877 (850) UBS Securities Australia Ltd Sell 7,874 (851) UBS Securities Australia Ltd Sell 7,874 (852) UBS Securities Australia Ltd Sell 7,275 (5,500) UBS Securities Australia Ltd Sell 1,787 (853) UBS Securities Australia Ltd Sell 1,782 (8,53) UBS Securities Australia Ltd Sell 1,202 (1,249) UBS Securities Australia Ltd Sell 1,055 (1,249) UBS Securities Australia Ltd Sell 1,056 (1,213) UBS Securities Australia Ltd Sell 1,056 (1,230) UBS Securities Australia Ltd Sell 1,066 (1,731) UBS Securities Australia Ltd Sell 1,068 (1,731) UBS Securities Australia Ltd Sell 1,068 (1,732) UBS Securities Australia Ltd Sell 1,068 (1,732) UBS Secur	3-May-14	UBS Securities Australia Ltd	Sell	336	(32)	Ordinary
UBS Securities Australia Ltd Sell 7.817 (860) UBS Securities Australia Ltd Sell 7.408 (861) UBS Securities Australia Ltd Sell 7.408 (815) UBS Securities Australia Ltd Sell 1.727 (862) UBS Securities Australia Ltd Sell 1.175 (12.49) UBS Securities Australia Ltd Sell 1.175 (12.29) UBS Securities Australia Ltd Sell 1.175 (12.29) UBS Securities Australia Ltd Sell 1.105 (12.29) UBS Securities Australia Ltd Sell 1.105 (12.20) UBS Securities Australia Ltd Sell 1.106 (1.209) UBS Securities Australia Ltd Sell 1.106 (1.201) UBS Securities Australia Ltd Sell 1.2072 (1.529) UBS Securities Australia Ltd Sell 1.2072 (1.751) UBS Securities Australia Ltd Sell 1.2072 (1.751) UBS Securities Australia Ltd Sell 1.2072 (1.751) UB	3-May-14	UBS Securities Australia Ltd	Sell	12,099	(1,331)	Ordinary
UBS Securities Australia Ltd Sell R.7408 (863) UBS Securities Australia Ltd Sell 8.754 (863) UBS Securities Australia Ltd Sell 2.775 (850) UBS Securities Australia Ltd Sell 4.527 (503) UBS Securities Australia Ltd Sell 11,782 (1.249) UBS Securities Australia Ltd Sell 11,782 (1.249) UBS Securities Australia Ltd Sell 11,005 (1.213) UBS Securities Australia Ltd Sell 11,006 (1.213) UBS Securities Australia Ltd Sell 1,405 (1.249) UBS Securities Australia Ltd Sell 1,609 (1.239) UBS Securities Australia Ltd Sell 1,600 (1.745) UBS Securities Australia Ltd Sell 1,600 (1.745) UBS Securities Australia Ltd Sell 1,745 (1.745) UBS Securities Australia Ltd Sell 6,712 (1.745) UBS Securities Australia Ltd Sell 6,712 (1.745) U	3-May-14	UBS Securities Australia Ltd	Sell	7,817	(860)	Ordinary
UBS Securities Australia Ltd Sell 8.754 (953) UBS Securities Australia Ltd Sell 7,405 (8,103) UBS Securities Australia Ltd Sell 11,762 (7,500) UBS Securities Australia Ltd Sell 11,762 (7,240) UBS Securities Australia Ltd Sell 11,762 (1,240) UBS Securities Australia Ltd Sell 11,762 (1,240) UBS Securities Australia Ltd Sell 1,656 (1,240) UBS Securities Australia Ltd Sell 1,656 (1,213) UBS Securities Australia Ltd Sell 1,656 (1,219) UBS Securities Australia Ltd Sell 1,669 (1,219) UBS Securities Australia Ltd Sell 1,785 (1,250) UBS Securities Australia Ltd Sell 1,785 (1,250) UBS Securities Australia Ltd Sell 1,785 (1,50) UBS Securities Australia Ltd Sell 1,785 (1,50) UBS Securities Australia Ltd Sell 1,787 (1,50) <td< td=""><td>3-May-14</td><td>UBS Securities Australia Ltd</td><td>Sel</td><td>7,817</td><td>(890)</td><td>Ordinary</td></td<>	3-May-14	UBS Securities Australia Ltd	Sel	7,817	(890)	Ordinary
UBS Securities Australia Ltd Sell 7,408 (815) UBS Securities Australia Ltd Sell 4,572 (3,500) UBS Securities Australia Ltd Sell 4,572 (3,500) UBS Securities Australia Ltd Sell 1,756 (1,249) UBS Securities Australia Ltd Sell 1,787 (802) UBS Securities Australia Ltd Sell 1,026 (1,249) UBS Securities Australia Ltd Sell 1,026 (1,213) UBS Securities Australia Ltd Sell 1,026 (1,75) UBS Securities Australia Ltd Sell 1,026 (1,752) UBS Securities Australia Ltd Sell 1,026 (1,752) UBS Securities Australia Ltd Sell 4,272 (473) UBS S	3-May-14	UBS Securities Australia Ltd	Sell	8,754	(696)	Ordinary
UBS Securities Australia Ltd Sell 4,572 (2,500) UBS Securities Australia Ltd Sell 4,572 (533) UBS Securities Australia Ltd Sell 11,762 (1,249) UBS Securities Australia Ltd Sell 11,762 (1,249) UBS Securities Australia Ltd Sell 11,623 (1,249) UBS Securities Australia Ltd Sell 1,653 (1,219) UBS Securities Australia Ltd Sell 1,053 (1,219) UBS Securities Australia Ltd Sell 1,053 (1,219) UBS Securities Australia Ltd Sell 1,063 (1,219) UBS Securities Australia Ltd Sell 1,063 (1,502) UBS Securities Australia Ltd Sell 1,063 (1,502) UBS Securities Australia Ltd Sell 1,063 (1,502) UBS Securities Australia Ltd Sell 1,068 (1,507) UBS Securities Australia Ltd Sell 1,068 (1,507) UBS Securities Australia Ltd Sell 2,136 (1,507) <	3-May-14	UBS Securities Australia Ltd	Seli	7,408	(815)	Ordinary
UBS Securities Australia Ltd Sell 4,572 (503) UBS Securities Australia Ltd Sell 11,782 (1,294) UBS Securities Australia Ltd Sell 11,583 (1,294) UBS Securities Australia Ltd Sell 11,573 (1,294) UBS Securities Australia Ltd Sell 1,0553 (1,205) UBS Securities Australia Ltd Sell 1,0553 (1,205) UBS Securities Australia Ltd Sell 1,0563 (1,205) UBS Securities Australia Ltd Sell 1,062 (1,205) UBS Securities Australia Ltd Sell 1,076 (1,528) UBS Securities Australia Ltd Sell 1,076 (1,528) UBS Securities Australia Ltd Sell 1,076 (1,528) UBS Securities Australia Ltd Sell 1,076 (1,758) UBS Securities Australia Ltd Sell 1,076 (1,758) UBS Securities Australia Ltd Sell 1,078 (1,507) UBS Securities Australia Ltd Sell 1,078 (1,507)	3-May-14	UBS Securities Australia Ltd	les	22,725	(2,500)	Ordinary
UBS Securities Australia Ltd Sell 11,762 (1,249) UBS Securities Australia Ltd Sell 11,762 (1,249) UBS Securities Australia Ltd Sell 11,352 (1,249) UBS Securities Australia Ltd Sell 10,953 (1,249) UBS Securities Australia Ltd Sell 11,026 (1,219) UBS Securities Australia Ltd Sell 1,063 (1,219) UBS Securities Australia Ltd Sell 1,045 (1,00) UBS Securities Australia Ltd Sell 1,046 (1,00) <td< td=""><td>3-May-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>4,572</td><td>(203)</td><td>Ordinary</td></td<>	3-May-14	UBS Securities Australia Ltd	Sell	4,572	(203)	Ordinary
UBS Securities Australia Ltd Sell 11,752 (1,249) UBS Securities Australia Ltd Sell 8,754 (663) UBS Securities Australia Ltd Sell 10,953 (1,249) UBS Securities Australia Ltd Sell 10,953 (1,213) UBS Securities Australia Ltd Sell 11,026 (1,213) UBS Securities Australia Ltd Sell 1,400 (1,519) UBS Securities Australia Ltd Sell 1,400 (1,519) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,836 (7,072) UBS Securities Australia Ltd Sell 1,639 (1,573) UBS Securities Australia Ltd Sell 1,040 (1,152) UBS Securities Australia Ltd Sell 1,040 (1,572) UBS Securities Australia Ltd Sell 1,040 (1,572) UBS Securities Australia Ltd Sell 4,272 (470) UB	3-May-14	UBS Securities Australia Ltd	Seil	999	(73)	Ordinary
UBS Securities Australia Ltd Sell 11,353 (1,249) UBS Securities Australia Ltd Sell 7,817 (860) UBS Securities Australia Ltd Sell 1,0553 (1,205) UBS Securities Australia Ltd Sell 1,026 (1,219) UBS Securities Australia Ltd Sell 1,081 (1,219) UBS Securities Australia Ltd Sell 1,081 (1,219) UBS Securities Australia Ltd Sell 1,081 (1,219) UBS Securities Australia Ltd Sell 1,082 (1,748) UBS Securities Australia Ltd Sell 1,088 (1,718) UBS Securities Australia Ltd Sell 1,045 (1,745) UBS Securities Australia Ltd Sell 1,045 (1,745) UBS Securities Australia Ltd Sell 1,045 (1,757) UBS Securities Australia Ltd Sell 1,045 (1,757) UBS Securities Australia Ltd Sell 1,045 (1,757) UBS Securities Australia Ltd Sell 2,136 (1,577) <t< td=""><td>3-May-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>11,762</td><td>(1,294)</td><td>Ordinary</td></t<>	3-May-14	UBS Securities Australia Ltd	Sell	11,762	(1,294)	Ordinary
UBS Securities Australia Ltd Sell 8,754 (963) UBS Securities Australia Ltd Sell 1,026 (1,213) UBS Securities Australia Ltd Sell 1,026 (1,213) UBS Securities Australia Ltd Sell 7,427 (817) UBS Securities Australia Ltd Sell 1,086 (1,219) UBS Securities Australia Ltd Sell 1,080 (1,219) UBS Securities Australia Ltd Sell 1,080 (1,202) UBS Securities Australia Ltd Sell 1,080 (1,020) UBS Securities Australia Ltd Sell 1,045 (1,07) UBS Securities Australia Ltd Sell 1,045 (1,07) UBS Securities Australia Ltd Sell 1,045 (1,00) UBS Securities Australia Ltd Sell 2,136 (1,00) UBS Se	3-May-14	UBS Securities Australia Ltd	Sell	11,353	(1.249)	Ordinary
UBS Securities Australia Ltd Sell 17,817 (860) UBS Securities Australia Ltd Sell 11,025 (1,205) UBS Securities Australia Ltd Sell 1,427 (817) UBS Securities Australia Ltd Sell 1,409 (1,219) UBS Securities Australia Ltd Sell 1,409 (1,529) UBS Securities Australia Ltd Sell 1,836 (702) UBS Securities Australia Ltd Sell 1,048 (702) UBS Securities Australia Ltd Sell 1,048 (702) UBS Securities Australia Ltd Sell 1,048 (702) UBS Securities Australia Ltd Sell 1,040 (702) UBS Securities Australia Ltd Sell 1,040 (702) UBS Securities Australia Ltd Sell 2,136 (7,507) UBS Securities Aus	3-May-14	UBS Securities Australia Ltd	Sell	8.754	(863)	Ordinary
UBS Securities Australia Ltd Sell 10,953 (1,205) UBS Securities Australia Ltd Sell 7,663 (843) UBS Securities Australia Ltd Sell 1,081 (1,219) UBS Securities Australia Ltd Sell 1,081 (1,219) UBS Securities Australia Ltd Sell 1,082 (1,219) UBS Securities Australia Ltd Sell 1,082 (1,745) UBS Securities Australia Ltd Sell 1,082 (1,745) UBS Securities Australia Ltd Sell 1,084 (1,755) UBS Securities Australia Ltd Sell 1,045 (1,155) UBS Securities Australia Ltd Sell 1,045 (1,157) UBS Securities Australia Ltd Sell 6,718 (7,67) UBS Securities Australia Ltd Sell 2,136 (2,507) UBS Securities Australia Ltd Sell 1,244 (1,129) UBS Securities Australia Ltd Sell 2,136 (7,57) UBS Securities Australia Ltd Sell 1,569 (7,507) U	3-May-14	UBS Securities Australia Ltd	Sell	7.817	(860)	Ordinary
UBS Securities Australia Ltd Sell 11,026 (1,213) UBS Securities Australia Ltd Sell 7,427 (817) UBS Securities Australia Ltd Sell 1,409 (1,519) UBS Securities Australia Ltd Sell 1,409 (1,528) UBS Securities Australia Ltd Sell 1,607 (1,528) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,649 (1,155) UBS Securities Australia Ltd Sell 1,645 (1,155) UBS Securities Australia Ltd Sell 1,045 (1,155) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 2,136 (1,507) UBS Securities Australia Ltd Sell 2,136 (1,507) UBS Securities Australia Ltd Sell 2,136 (1,507) UBS Securities Australia Ltd Sell 1,174 (1,507) U	3-May-14	UBS Securities Australia Ltd	Sell	10.953	(1 205)	Ordinacy
UBS Securities Australia Ltd Sell 7,427 (813) UBS Securities Australia Ltd Sell 7,427 (814) UBS Securities Australia Ltd Sell 1,083 (1,299) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,882 (1,745) UBS Securities Australia Ltd Sell 1,886 (1,745) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 6,972 (470) UBS Securities Australia Ltd Sell 6,972 (470) UBS Securities Australia Ltd Sell 2,136 (1,571) UBS Securities Australia Ltd Sell 1,544 (1,132) UBS Securities Australia Ltd Sell 1,544 (1,132) UBS Securities Australia Ltd Sell 1,509 (1,102) UBS Securi	3-Mav-14	IUBS Securities Australia Ltd	la _S	11.026	(1 213)	Ordinary
UBS Securities Australia Ltd Sell 1,663 (843) UBS Securities Australia Ltd Sell 1,081 (1,219) UBS Securities Australia Ltd Sell 1,2072 (1,328) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,836 (7,745) UBS Securities Australia Ltd Sell 1,886 (7,745) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 6,718 (7,787) UBS Securities Australia Ltd Sell 6,718 (7,787) UBS Securities Australia Ltd Sell 2,738 (7,677) UBS Securities Australia Ltd Sell 2,136 (7,577) UBS Securities Australia Ltd Sell 1,134 (1,1292) UBS Securities Australia Ltd Sell 1,134 (1,1292) UBS Securities Australia Ltd Sell 1,244 (1,1292) <t< td=""><td>3-May-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>7 427</td><td>(817)</td><td>Ordinary</td></t<>	3-May-14	UBS Securities Australia Ltd	Sell	7 427	(817)	Ordinary
UBS Securities Australia Ltd Sell 1,1081 (1,219) UBS Securities Australia Ltd Sell 1,409 (155) UBS Securities Australia Ltd Sell 1,2072 (1,239) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 15,862 (1,745) UBS Securities Australia Ltd Sell 10,683 (1,745) UBS Securities Australia Ltd Sell 1,048 (1,755) UBS Securities Australia Ltd Sell 1,048 (1,175) UBS Securities Australia Ltd Sell 1,048 (1,175) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 2,136 (1,507) UBS Securities Australia Ltd Sell 2,136 (1,507) UBS Securities Australia Ltd Sell 2,136 (1,507) UBS Securities Australia Ltd Sell 1,504 (1,101) UBS Securities Australia Ltd Sell 2,136 (1,507)	3-May-14	UBS Securities Australia I td	- V	7,663	(8/3)	Craining
UBS Securities Australia Ltd Sell 1,000 (1,528) UBS Securities Australia Ltd Sell 1,002 (1,328) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,882 (1,745) UBS Securities Australia Ltd Sell 1,688 (1,745) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 6,972 (1,757) UBS Securities Australia Ltd Sell 6,972 (1,757) UBS Securities Australia Ltd Sell 2,136 (1,577) UBS Securities Australia Ltd Sell 1,569 (1,507) UBS Securities Australia Ltd Sell 1,509 (1,507) UBS Securities Australia Ltd Sell 1,509 (1,507) UBS Securities Australia Ltd Sell 1,509 (1,109) UBS Securities Australia Ltd Sell 1,509 (1,109)	3-Mav-14	UBS Securities Australia 11d	ll de	11.081	(1 210)	Ordinary
UBS Securities Australia Ltd Sell 1,2,72 (1,120) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,836 (7,02) UBS Securities Australia Ltd Sell 15,862 (7,02) UBS Securities Australia Ltd Sell 16,803 (7,00) UBS Securities Australia Ltd Sell 1,045 (1,15) UBS Securities Australia Ltd Sell 6,718 (7,73) UBS Securities Australia Ltd Sell 6,718 (7,73) UBS Securities Australia Ltd Sell 6,718 (7,73) UBS Securities Australia Ltd Sell 2,136 (7,57) UBS Securities Australia Ltd Sell 1,569 (7,507) UBS Securities Australia Ltd Sell 1,569 (7,507) UBS Securities Australia Ltd Sell 1,744 (1,292) UBS Securities Australia Ltd Sell 1,744 (1,292) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Se	8-May-14	IBS Securities Australia 11d	3 3	1 409	(1513)	Ordinary
UBS Securities Australia Ltd Sell 1,326 (1,329) UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,836 (1,045) UBS Securities Australia Ltd Sell 1,045 (1,050) UBS Securities Australia Ltd Sell 1,045 (1,15) UBS Securities Australia Ltd Sell 6,718 (739) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 2,136 (1,577) UBS Securities Australia Ltd Sell 2,136 (1,577) UBS Securities Australia Ltd Sell 1,544 (1,527) UBS Securities Australia Ltd Sell 1,344 (1,527) UBS Securities Australia Ltd Sell 1,344 (1,222) UBS Securities Australia Ltd Sell 1,344 (1,227) UBS Secur	May 17	JOS Committee Ametralia Inch	= 20	CEO C.	(000 1)	S C C
UBS Securities Australia Ltd Sell 1,836 (202) UBS Securities Australia Ltd Sell 1,836 (1,745) UBS Securities Australia Ltd Sell 11,880 (1,745) UBS Securities Australia Ltd Sell 1,645 (1,175) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 6,972 (767) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 2,136 (250) UBS Securities Australia Ltd Sell 2,136 (1,507) UBS Securities Australia Ltd Sell 1,134 (1,132) UBS Securities Australia Ltd Sell 1,309 (1,107) UBS Securities Australia Ltd Sell 1,309 (1,109) UBS Securities Australia Ltd Sell 1,309 (1,109) UBS Securities Australia Ltd Sell 1,309 (1,100) UBS Sec	1 - Kan 1 -	Upo Securities Australia Liu	200	77077	(1,328)	Ordinary
UBS Securities Australia Ltd Sell 1,836 (702) UBS Securities Australia Ltd Sell 15,825 (1,745) UBS Securities Australia Ltd Sell 1,045 (1,15) UBS Securities Australia Ltd Sell 1,045 (1,15) UBS Securities Australia Ltd Sell 6,718 (739) UBS Securities Australia Ltd Sell 6,718 (739) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 2,136 (55) UBS Securities Australia Ltd Sell 1,569 (750) UBS Securities Australia Ltd Sell 1,569 (750) UBS Securities Australia Ltd Sell 1,569 (750) UBS Securities Australia Ltd Sell 1,344 (1,507) UBS Securities Australia Ltd Sell 1,344 (1,507) UBS Securities Australia Ltd Sell 1,344 (1,292) UBS Securities Australia Ltd Sell 1,344 (1,202) UBS Securities A	- Ividy-14	Obs securities Australia Ltd		08,1	(707)	Ordinary
UBS Securities Australia Ltd Sell 15,862 (1,43) UBS Securities Australia Ltd Sell 18,800 (20,000) UBS Securities Australia Ltd Sell 1,045 (1,15) UBS Securities Australia Ltd Sell 1,048 (1,15) UBS Securities Australia Ltd Sell 6,718 (739) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 2,136 (1,577) UBS Securities Australia Ltd Sell 1,594 (1,575) UBS Securities Australia Ltd Sell 5,382 (1,577) UBS Securities Australia Ltd Sell 5,382 (1,575) UBS Securities Australia Ltd Sell 3,136 (1,522) UBS Securities Australia Ltd Sell 3,136 (1,00) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Sec	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	UBS Securities Australia LIO		958,1	(202)	Ordinary
UBS Securities Australia Ltd Sell 181,800 (20,000) UBS Securities Australia Ltd Sell 1,045 (1,175) UBS Securities Australia Ltd Sell 6,972 (1,175) UBS Securities Australia Ltd Sell 6,972 (767) UBS Securities Australia Ltd Sell 6,972 (767) UBS Securities Australia Ltd Sell 2,136 (235) UBS Securities Australia Ltd Sell 2,136 (1,577) UBS Securities Australia Ltd Sell 1,569 (1,577) UBS Securities Australia Ltd Sell 1,544 (1,134) UBS Securities Australia Ltd Sell 5,300 (1,577) UBS Securities Australia Ltd Sell 5,304 (1,527) UBS Securities Australia Ltd Sell 1,309 (1,102) UBS Securities Australia Ltd Sell 1,309 (1,102) UBS Securities Australia Ltd Sell 1,309 (1,101) UBS Securities Australia Ltd Sell 1,309 (1,101) U		UBS Securities Australia Ltd	lia C	798'51	(1,/45)	Ordinary
Uses Securities Australia Ltd Sell 11,045 (1,175) Uses Securities Australia Ltd Sell 1,045 (1,175) Uses Securities Australia Ltd Sell 1,045 (1,175) Uses Securities Australia Ltd Sell 6,718 (739) Uses Securities Australia Ltd Sell 4,272 (470) Uses Securities Australia Ltd Sell 2,382 (52) Uses Securities Australia Ltd Sell 1,569 (1,507) Uses Securities Australia Ltd Sell 1,369 (1,507) Uses Securities Australia Ltd Sell 1,369 (1,507) Uses Securities Australia Ltd Sell 1,344 (1,322) Uses Securities Australia Ltd Sell 1,344 (1,322) Uses Securities Australia Ltd Sell 1,744 (1,232) Uses Securities Australia Ltd Sell 1,000 (1,10) Uses Securities Australia Ltd Sell 1,000 (1,10) Uses Securities Australia Ltd Sell 1,000 (1,10)	-IMay-14	UBS Securities Australia Ltd	Sell	16	(10)	Ordinary
UBS Securities Australia Ltd Sell 10,681 (1,175) UBS Securities Australia Ltd Sell 6,718 (175) UBS Securities Australia Ltd Sell 6,712 (773) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 1,362 (255) UBS Securities Australia Ltd Sell 1,364 (1,314) UBS Securities Australia Ltd Sell 1,364 (1,314) UBS Securities Australia Ltd Sell 5,136 (1,507) UBS Securities Australia Ltd Sell 1,304 (1,214) UBS Securities Australia Ltd Sell 1,304 (1,207) UBS Securities Australia Ltd Sell 1,309 (1,10) UBS Securities Australia Ltd Sell 1,309 (1,10) UBS Securities Australia Ltd Sell 1,309 (1,10) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Securit	S-May-14	UBS Securities Australia Ltd	Sell	181,800	(20,000)	Ordinary
UBS Securities Australia Ltd Sell 1,045 (115) UBS Securities Australia Ltd Sell 6,772 (767) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 2,136 (235) UBS Securities Australia Ltd Sell 2,136 (235) UBS Securities Australia Ltd Sell 11,544 (1,207) UBS Securities Australia Ltd Sell 13,699 (1,507) UBS Securities Australia Ltd Sell 11,744 (1,292) UBS Securities Australia Ltd Sell 11,744 (1,292) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Securi	S-May-14	UBS Securities Australia Ltd	Sell	10,681	(1,175)	Ordinary
UBS Securities Australia Ltd Sell 6,718 (739) UBS Securities Australia Ltd Sell 4,272 (4707) UBS Securities Australia Ltd Sell 4,272 (4707) UBS Securities Australia Ltd Sell 2,136 (51) UBS Securities Australia Ltd Sell 2,136 (7507) UBS Securities Australia Ltd Sell 1,544 (1,507) UBS Securities Australia Ltd Sell 6,863 (7,507) UBS Securities Australia Ltd Sell 6,863 (7,507) UBS Securities Australia Ltd Sell 94,909 (1,041) UBS Securities Australia Ltd Sell 3,136 (7,522) UBS Securities Australia Ltd Sell 3,136 (7,007) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Secur	3-May-14	UBS Securities Australia Ltd	Sell	1,045	(115)	Ordinary
UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 2,136 (355) UBS Securities Australia Ltd Sell 2,136 (235) UBS Securities Australia Ltd Sell 13,629 (1,527) UBS Securities Australia Ltd Sell 11,544 (1,314) UBS Securities Australia Ltd Sell 51,304 (1,527) UBS Securities Australia Ltd Sell 11,744 (1,314) UBS Securities Australia Ltd Sell 11,744 (1,222) UBS Securities Australia Ltd Sell 11,744 (1,222) UBS Securities Australia Ltd Sell 1,909 (1,10) UBS Securities Australia Ltd Sell 1,909 (1,10) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Securities Australia Ltd Sell 91 (1,0) UBS Securities Australia Ltd Sell 91 (1,0) UBS Securitie	3-May-14	UBS Securities Australia Ltd	PS	6,718	(739)	Ordinary
UBS Securities Australia Ltd Sell 4,272 (470) UBS Securities Australia Ltd Sell 2,136 (51) UBS Securities Australia Ltd Sell 2,138 (262) UBS Securities Australia Ltd Sell 13,699 (7,507) UBS Securities Australia Ltd Sell 11,944 (1,134) UBS Securities Australia Ltd Sell 6,663 (7,551) UBS Securities Australia Ltd Sell 9,909 (1,102) UBS Securities Australia Ltd Sell 11,744 (1,1292) UBS Securities Australia Ltd Sell 3,136 (10,10) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Austra	8-May-14	UBS Securities Australia Ltd	Sell	6,972	(767)	Ordinary
UBS Securities Australia Ltd Sell 464 (51) UBS Securities Australia Ltd Sell 2,136 (225) UBS Securities Australia Ltd Sell 1,569 (1,507) UBS Securities Australia Ltd Sell 1,544 (1,507) UBS Securities Australia Ltd Sell 5,130 (1,507) UBS Securities Australia Ltd Sell 5,130 (1,507) UBS Securities Australia Ltd Sell 5,130 (1,507) UBS Securities Australia Ltd Sell 3,136 (1,502) UBS Securities Australia Ltd Sell 1,909 (1,101) UBS Securities Australia Ltd Sell 1,909 (1,101) UBS Securities Australia Ltd Sell 1,900 (1,101) UBS Securities Australia Ltd Sell 1,900 (1,101) UBS Securities Australia Ltd Sell 9,999 (1,100) UBS Securities Australia Ltd Sell 9,999 (1,100) UBS Securities Australia Ltd Sell 9,999 (1,100) UBS S	3-May-14	UBS Securities Australia Ltd	Sell	4,272	(470)	Ordinary
UBS Securities Australia Ltd Sell 2,136 (235) UBS Securities Australia Ltd Sell 13,822 (282) UBS Securities Australia Ltd Sell 11,544 (1,314) UBS Securities Australia Ltd Sell 11,744 (1,222) UBS Securities Australia Ltd Sell 11,744 (1,222) UBS Securities Australia Ltd Sell 1,909 (1,10) UBS Securities Australia Ltd Sell 1,909 (1,10) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Securities Australia Ltd Sell 91 (1,0) UBS Securities Aus	5-May-14	UBS Securities Australia Ltd	Sell	464	(51)	Ordinary
UBS Securities Australia Ltd Sell 1,382 (562) UBS Securities Australia Ltd Sell 1,3699 (1,507) UBS Securities Australia Ltd Sell 1,3699 (1,507) UBS Securities Australia Ltd Sell 6,863 (7,541) UBS Securities Australia Ltd Sell 94,909 (1,241) UBS Securities Australia Ltd Sell 11,744 (1,292) UBS Securities Australia Ltd Sell 3,136 (10,411) UBS Securities Australia Ltd Sell 1,000 (1,10) UBS Securities Australia Ltd Sell 91 (1,00) UBS Securities	5-May-14	UBS Securities Australia Ltd	Sell	2,136	(235)	Ordinary
UBS Securities Australia Ltd Sell 13,699 (1,507) UBS Securities Australia Ltd Sell 11,944 (1,314) UBS Securities Australia Ltd Sell 6,863 (755) UBS Securities Australia Ltd Sell 94,909 (10,441) UBS Securities Australia Ltd Sell 94,909 (10,441) UBS Securities Australia Ltd Sell 3,136 (3,23) UBS Securities Australia Ltd Sell 1,909 (2,10) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd	3-May-14	UBS Securities Australia Ltd	Sell	2,382	(262)	Ordinary
UBS Securities Australia Ltd Sell 11,944 (1,314) UBS Securities Australia Ltd Sell 6,863 (755) UBS Securities Australia Ltd Sell 51,304 (1,641) UBS Securities Australia Ltd Sell 94,909 (10,441) UBS Securities Australia Ltd Sell 1,744 (1,222) UBS Securities Australia Ltd Sell 3,136 (345) UBS Securities Australia Ltd Sell 1,909 (210) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd <t< td=""><td>-May-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>13,699</td><td>(1.507)</td><td>Ordinary</td></t<>	-May-14	UBS Securities Australia Ltd	Sell	13,699	(1.507)	Ordinary
UBS Securities Australia Ltd Sell 6,863 (755) UBS Securities Australia Ltd Sell 5,304 (6,644) UBS Securities Australia Ltd Sell 94,909 (10,441) UBS Securities Australia Ltd Sell 11,744 (1,292) UBS Securities Australia Ltd Sell 3,136 (733) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 91 (1,00)	-May-14	UBS Securities Australia Ltd	IjaS	11.944	(1.314)	Ordinary
UBS Securities Australia Ltd Sell 51,304 (5,644) UBS Securities Australia Ltd Sell 94,909 (10,441) UBS Securities Australia Ltd Sell 1,774 (1,292) UBS Securities Australia Ltd Sell 3,136 (345) UBS Securities Australia Ltd Sell 1,509 (2,10) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 91 (10) Bescurities Australia Ltd Sell 91 (10)	-May-14	UBS Securities Australia Ltd	les	6.863	(752)	Ordinary
UBS Securities Australia Ltd Sell 94,509 (10,441) UBS Securities Australia Ltd Sell 11,744 (1,222) UBS Securities Australia Ltd Sell 6,663 (733) UBS Securities Australia Ltd Sell 1,909 (210) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 91 (10)	-Mav-14	UBS Securities Australia Ltd	٩	202 13	(5,644)	Ordinary
UBS Securities Australia Ltd Sell 11,744 (1,592) UBS Securities Australia Ltd Sell 6653 (733) UBS Securities Australia Ltd Sell 3,136 (210) UBS Securities Australia Ltd Sell 1,909 (210) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 91 (10)	-Mav-14	UBS Securities Australia Ltd	le S	000 00	(10,441)	Ordinary Ordinary
UBS Securities Australia Ltd Sell 6,663 (7,133) UBS Securities Australia Ltd Sell 3,136 (345) UBS Securities Australia Ltd Sell 1,509 (2 10) UBS Securities Australia Ltd Sell 91 (10)	-May-14	IIBS Sequities Australia I to	130	11 744	(1 202)	Ordinary
UBS Securities Australia Ltd Sell 9,000 (139) UBS Securities Australia Ltd Sell 1,909 (210) UBS Securities Australia Ltd Sell 1,000 (110) UBS Securities Australia Ltd Sell 91 (10)	May-14	LIBS Securities Australia Ltd	100	632.3	(267)	Cidinary
UBS Securities Australia Ltd Sell 1,500 (2.10) UBS Securities Australia Ltd Sell 1,000 (1.10) UBS Securities Australia Ltd Sell 91 (1.0) UBS Securities Australia Ltd Sell 91 (1.100)	1	IRS Securities Australia Ltd	J. J.	2 126	(737)	Ordinary
USS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 93 (1,100) UBS Securities Australia Ltd Sell 93 (1,100) UBS Securities Australia Ltd Sell 1,773 (195)		LIPS Countries Australia Ltd	E 0	0001	(343)	Ordinary
UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 91 (100) UBS Securities Australia Ltd Sell 91 (100) UBS Securities Australia Ltd Sell 93 (100) UBS Securities Australia Ltd Sell 1773 (195)	-	Obs securities Australia Ltd	i Seli	606';	(012)	Ordinary
UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 91 (1,0) UBS Securities Australia Ltd Sell 9,999 (1,10) UBS Securities Australia Ltd Sell 9,999 (1,10) UBS Securities Australia Ltd Sell 9,999 (1,10)	- 1	UBS Securities Australia Ltd	Sell	1,000	(0110)	Ordinary
UBS Securities Australia Ltd Sell 91 (10) UBS Securities Australia Ltd Sell 9,999 (1,100) UBS Securities Australia Ltd Sell 9,999 (1,100) UBS Securities Australia Ltd Sell 9,999 (1,00)	-May-14	UBS Securities Australia Ltd	Sell	91	(10)	Ordinary
UBS Securities Australia Ltd Sell (1,00) UBS Securities Australia Ltd Sell (1,00) UBS Securities Australia Ltd Sell (1,00)	-May-14	UBS Securities Australia Ltd	Sell	91	(10)	Ordinary
UBS Securities Australia Ltd Sell (10)	May-14	UBS Securities Australia Ltd	Jes	666'6	(1,100)	Ordinary
[UBS Securities Australia (td Sell 1,773 (195)	-May-14	UBS Securities Australia Ltd	Sell	- 66	(10)	Ordinary
	-Mav-14					

07-Mav-14	UBS AG London Branch	ll aS	8,	(2)	Ordinary
37-May-14	UBS AG London Branch	Sell	2,332	(26	Ordinary
07-May-14	UBS AG London Branch	Sell	8,375		
07-May-14	UBS AG London Branch	Sell	2,023		
07-May-14	UBS AG London Branch	Sell	3,395		
07-May-14	UBS AG London Branch	Sell	11,779		Ordinary
07-May-14	UBS AG London Branch	Sell	6,094		
07-May-14	UBS AG London Branch	Sell	2,146	(241)	Ordinary
07-May-14	UBS AG London Branch	Seff	704		Ordinary
07-May-14	UBS AG London Branch	Sell	80,403	(9,034)	
07-May-14	UBS AG London Branch	les.	794,603	L	Ordinary
7-Mav-14	UBS AG London Branch	Buy	788 348		
07-May-14	HS Contrition Australia 14d	3	700,1		Crainery
07 May 14	100 Contraint Australia Ltd		105,1	ľ	Ciclinary
1 14 14	COO Securities Australia Lite	and .	05,367		Ordinary
ı	Obs Securities Australia Lid	ja j	/66'07	(3,030)	Ordinary
- 1	UBS Securities Australia Ltd	Buy	30,945		Ordinary
- 1	UBS Securities Australia Ltd	Self	37,520		Ordinary
07-May-14	UBS Securities Australia Ltd	Buy	788,348		Ordinary
•	UBS Securities Australia Ltd	Buv	89.537		Ordinary
07-May-14	UBS Securities Australia Ltd	les.	383		Ordinary
07-May-14	UBS Securities Australia Ltd	J.	109 985	(1)	
7-May-14	LIBS Wealth Management Australia (1d	S. S.	11 270		Ordinan
07-May-14	BS Wealth Management Australia 14	B. D.	2006		
07 May 14	100 World Management Australia 144	000	מיני לי		
7-IVIdy-14	Obs Wedith Management Australia Ltd	- X	13,235	(1,480)	
U/-May-14	UBS Wealth Management Australia Ltd	- N	94/		
U/-May-14	UBS securities Australia Ltd	Seil	151	(17)	Ordinary
07-May-14	UBS Securities Australia Ltd	Buy	3,747	421	Ordinary
7-May-14	UBS Securities Australia Ltd	Buy	25,881		Ordinary
7-May-14	UBS Securities Australia Ltd	auy 8	525		Ordinary
27-Mav-14	UBS Securities Australia Ltd	Buv	383		Ordinary
07-May-14	I IBS Securities Australia I td	Bir	244 519	27.6	Ordinan
02 hAm 14	I DO Contribut Australia 144	100	C10/113		Ciciliary
/-lvidy- 4	ODS SECULIFIES AUSTRAINE LIQ	Ode)	1,113		Ordinary
/-May-14	UBS Securities Australia Ltd	Buy	17,195		Ordinary
7-May-14	UBS Securities Australia Ltd	Buy	4,281	481	Ordinary
7-May-14	UBS Securities Australia Ltd	Buy	392	77	Ordinary
07-May-14	UBS Securities Australia Ltd	8cv	2,234	2	Ordinary
07-May-14	UBS Securities Australia Ltd	Beav	2 029		Ordinary
7-May-14	I IRS Securities Australia 14d	B. v.	0.452	1 062	Ordinan
07 1420, 14	1 30 Consisting August 1 and	9.5	2,422	200,1	O O
,	DIS SECURIOR CONTRACTOR	600	151,5	+00	Ciditary
07-Ividy-14	ODS SECURITIES AUSTRAINE LIU	and	45	0	Ordinary
0/-May-14	UBS Securities Australia Ltd	Buy	5,491	617	Ordinary
- 1	UBS Securities Australia Ltd	Buy	53		Ordinary
	UBS Securities Australia Ltd	Buy	338	38	Ordinary
ı	UBS Securities Australia Ltd	Buy	1.905	2	Ordinary
7-Mav-14	UBS Securities Australia Ltd	Buy	54 557	9	Ordinary
	I BS Societies Australia 14d	Bury	190 23		Ordinary
1	Securities Adalasia Etd	ina.	102,66		Cidinary
-	UBS Securities Australia Ltd	Buy	71,86/		Ordinary
	UBS Securities Australia Ltd	Buy	48,799	5,483	Ordinary
	UBS Securities Australia Ltd	Buy	18	2	Ordinary
07-Mav-14	UBS Securities Australia Ltd	Buv	6.052	089	Ordinary
1	I IRS Societies Australia Ltd	Bany	10 503	1 100	Ordinany
Т	CO CONTROL CON	ou,	200,01	001,1	Olumai y
07-May-14	UBS Securities Australia Ltd	Buy	134	15	Ordinary
	UBS Securities Australia Ltd	Buy	2,741	308	Ordinary
07-May-14	UBS Securities Australia Ltd	Buy	134	15	Ordinary
	UBS Securities Australia Itd	Riv	12 754		Ordinary
1	LIBS Securities Australia 14d	Burr	V30 C	976	Ordinary
-	DDC Countries Australia Ltd	, in	100,0		Cidilary
1	ODD Securities Australia and	- And	lc1,c	900	Ordinary
- 1	UBS Securities Australia Ltd	Buy	53	٥	Ordinary
/-May-14	UBS Securities Australia Ltd	Buy	53	9	Ordinary
- 1	UBS Securities Australia Ltd	Buy	214		Ordinary
	UBS Securities Australia Ltd	Buy	8,179	919	Ordinary
07-May-14	UBS Securities Australia ttd	Buv	1 237	139	Ordinary
07-May-14	I IRS Securities Australia 11d	Ruy	125	10	Ordinary
+I-klah-l	ODD Deculities Australia ciu	buy	671		Granaly
7-May-14	UBS Securities Australia Ltd	Buy	31,746	3,567	Ordinary
07-May-14	UBS Securities Australia Ltd	Buy	1,050	118	Ordinary
Г	18S Sequities Australia Ltd	20,000	3 151	V26	Ordinan
Т	טום מבתווויבים שתפיו מוומ	han	2.3	500	Cicinary
07-May-14	UBS Securities Australia Ltd	Buy	436	49	Ordinary
7	מס מבית וויכז ניימיים בים	food	1	}	5

TI ON ON	Opp Securities Australia Lita				
8-May-14	UBS Securities Australia Ltd	Sell	127	(14)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	2,863	(315)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	1,626	(179)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	1,626	(179)	Ordinary
08-May-14	UBS Securities Australia Ltd	Seli	1,072	(118)	Ordinary
8-May-14	UBS Securities Australia Ltd	Sell	182	(20)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	745	(82)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	3,600	(96E)	Ordinary
-May-14	UBS Securities Australia Ltd	Seli	482	(53)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	163	(84)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	255	(28)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	45	(5)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	1 973	(710)	Ordinary
08-May-14	UBS Securities Australia 11d	ay	10901	(120)	Ordinan
08-May-14	URS Securities Australia 11d	Sell	4 372	(481)	Ordinary
D8-54-34-14	I IRS Societies Australia 14d	130	100	(101)	Craining
D8-5434-14	I IBC Cocumities Australia (+4	Coll	1100	(01)	in in in
DS-May 14	I DC Countries Australia Ltd	ii-ac	1,100	(177)	Ordinary
N Nov. 17	1 IDC Cocurities Australia Ltd	1100	CF2	(17)	Ordinary
S May 1	1 IDC Countries Australia Ltd	lec les	6/0	(/4/)	Crainary
28-May-14	1 185 Securities Australia 14d	Sell		(C)	Ordinary
S May 14	LIDS Cocurities Australia Ltd	liec Sell	0 5	200	Cromary
10 May 14	LIDS Contribution Australia Ltd	liac liac	100	(01)	Oralnary
May-14	Upo Securities Australia Ltd	liec liec	-67	(35)	Orginary
00-May-14	UBS Securities Australia Ltd	Sell	810'-	(711)	Ordinary
00-May-14	Upo Securities Australia Liu	Sell Sell	3,218	(354)	Ordinary
May-14	Ups Securities Australia Ltd	llac .		(OL)	Ordinary
00-Mdy-14	UDS Securities Australia LIG	nec .	1/3	(08)	Ordinary
00-Mdy-14	Obs securities Australia Ltd	Sell	404	(10)	Ordinary
00-May-14	UBS Securities Australia Ltd	Sell	959'F	(400)	Ordinary
00-ivldy-14	Obs Securities Australia Ltd	ne ne ne ne ne ne ne ne ne ne ne ne ne n	<u></u>	(01)	Ordinary
US-May-14	UBS Securities Australia Ltd	Sell	1,22,1	(135)	Ordinary
-May- 14	UBS Securities Australia Ltd	Sell	1,218	(134)	Ordinary
J8-May-14	UBS Securities Australia Ltd	Sell	22,182	(2,443)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	5,866	(646)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	12,058	(1,328)	Ordinary
8-May-14	UBS Securities Australia Ltd	Sell	354	(36)	Ordinary
8-May-14	UBS Securities Australia Ltd	Sell	12,467	(1,373)	Ordinary
38-May-14	UBS Securities Australia Ltd	lleS.	136	(15)	Ordinary
38-May-14	UBS Securities Australia Ltd	3,8	18.160	(2.000)	Ordinary
38-May-14	UBS Securities Australia Ltd	ll-Sell	1.816	(200)	Ordinary
8-Mav-14	UBS Securities Australia Ltd	as	22,137	(2 438)	Ordinary
-May-14	I IRS Securities Australia I to	=======================================	254	(80)	Ordinary
OB-May-14	LIBC Securities Australia Ltd	P	10200	(1,000)	Ordinary
DR-May-14	I IRC Countries Australia Ltd	lles.	CC1 NC	(020,1)	Ordinary
00 May 14	LIDE Countries Australia Ltd	Dr	2C1,1FC	(907)	Ordinary
-inday-	מושוקה אחזורות	Į,	777'4	(405)	Ordinary
J8-May-14	UBS Securities Australia Ltd	- Ne	681,1	(151)	Ordinary
US-May-14	UBS Securities Australia Ltd	- Xell	065	(92)	Ordinary
-May-14	UBS Securities Australia Ltd	Seil	5,3/5	(265)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	1,126	(124)	Ordinary
08-May-14	UBS Securities Australia Ltd	Setl	10,342	(1,139)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	8,245	(806)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	989	(65)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	10,288	(1,133)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	2,152	(237)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	4.222	(465)	Ordinary
08-May-14	UBS Securities Australia 11d	و	11,731	(1 292)	Ordinary
-May-14	UBS Securities Australia Ltd	35	13.121	(1.445)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	9.915	(1 092)	Ordinary
08-May-14	UBS Securities Australia Hd	lleS	15.064	(1,659)	Ordinary
08-May-14	UBS Securities Australia Ltd	3	7 300	(808)	Ordinary
08-May-14	I IBC Con riting Australia 14d	Cell	7,300	(500)	Ordinary
20-tylay-14	Under Securities Australia Ltd	liec 3	118,0	(100)	Ordinary
May 14	UBS Securities Australia Ltd	ii Neii	45,400	(000,5)	Ordinary
00-May-14	Obs securities Australia Ltd	nen o	500,00	(1,004)	Orginary
US-May-14	UBS Securities Australia Ltd	Sell	002'/	(86/)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	303 C		
	and an arrangement of the second		nec'c	(365)	Ordinary

			777	(82)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	7//	3	
08-May-14	UBS Securities Australia Ltd	Self	16	(10)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	1,344	(148)	Ordinary
38-May-14	UBS Securities Australia Ltd	ll-8S	708	(28)	Ordinary
08-May-14	UBS Securities Australia Ltd	le?	82	Ó	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	2 125	(734)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	3.232	(356)	Ordinary
8-May-14	UBS Securities Australia Ltd	Sell	1.244	(137)	Ordinary
08-May-14	UBS Securities Australia Ltd	38	681	(75)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	436	(48)	Ordinary
08-May-14	UBS Securities Australia Ltd	95	22.055	(2.429)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	22.782	(2.509)	Ordinary
8-Mav-14	UBS Securities Australia I td	95	24 580	(2 707)	Ordinary
08-May-14	UBS Securities Australia Hd	3	2 161	(338)	Ordinary
08-May-14	I IRS Securities Australia I I d	3	101,01	(2.114)	Ordings
18-May-14	LIBS Securities Australia 144	ll o	051,51	(201)	Ciginary
38-May-14	LIBS Securities Australia 11d	195	2,370	(10)	Ordinary
08-May-14	LIBS Securities Australia 14d	Coll	16	(10)	Ordingery
8-May-14	LIRS Securities Australia 144	Coll	Pr o	(10)	Ordinary
8-May-14	LIBS Securities Australia Ltd	To D	0	000	Ordinary
08-May-14	UBS Securities Australia I td	3	5	(01)	Ordinary
28-May-14	LIBS Securities Australia 11d	3	5 6	6 6	Ordinary
18-May-14	I IBC Contribute Australia 14d	les.	5	(01)	Ordinary
De May 14	TIPS Counties Australia Ltd	nec les	ñ 8	(0)	Ordinary
8.May-17	URC Countries Australia 144	120	0000	(1000)	Ordinary
28-May-14	I RS Securities Australia 11d	las las	2000	(000,1)	Ordinary
N-4-14	I IRC Socienting Australia 14d	100	3306	(876)	Ordings
18-May-14	LIRS Securities Australia 11d	la S	155	(121)	Ordinally
18-May-14	1 MS Societies Australia 14d	Coll	30 6	(900)	Ordinary
08-May-14	LIBS Securities Australia and	===	2,855	(755)	Ordinary
18-May-14	1 195 Contrition Australia 14d	100	13 034	(50.7)	Cidinal
8 May 14	LIBS Contrition Australia 1+0	100	136,7	(4.04)	Ordinal y
08-May-14	LIBS Securities Australia 11d	(e)	1 053	(317)	Ordinary
A-May-14	I IBC Countries Australia (+d	100	1 063	315	Ordinary.
8-May-14	LIBS Counties Australia Ltd	ll-S	500,1	(110)	Ordinary
OB May 14	TIPO Contribing Australia 140	= 0	200,	(911)	S C C
10 M	TIPS Committee Australia Ltd	100	OEC.C	(4/5)	Guilaly
00-Ividy-14	Obs Securities Australia Liti	Jac .	5	(01.)	Ordinary
to I - KPIMI-G	UBS Securities Australia Ltd	Sell	3,169	(349)	Ordinary
No-May-14	Obs securities Australia Ltd	liac .	69/'9	(/45)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	4,041	(445)	Ordinary
38-May-14	UBS Securities Australia Ltd	Setl	12,412	(1,367)	Ordinary
38-May-14	UBS Securities Australia Ltd	Setl	1,053	(116)	Ordinary
8-May-14	UBS Securities Australia Ltd	Sell	889'6	(1,067)	Ordinary
38-May-14	UBS Securities Australia Ltd	Seil	372	(41)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	13,584	(1,496)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	14,010	(1,543)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	19,872	(2,191)	Ordinary
8-May-14	UBS Securities Australia Ltd	Sell	72	(3)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	7,818	(862)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	4,819	(531)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	653	(72)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	2,133	(232)	Ordinary
08-May-14	UBS Securities Australia Ltd	Seli	7,818	(862)	Ordinary
3-May-14	UBS Securities Australia Ltd	Se	2.133	(232)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	6 788	(748)	Ordinary
8-May-14	UBS Securities Australia Ltd	100	10.699	(1179)	Ordinary
08-May-14	I IBS Securities Australia I to	<u> </u>	21 514	(0.27)	Ordinary
08-May-14	UBS Securities Australia Ltd	Jel.	754	(8/2)	Ordinary
8-May-14	I BC Counties Australia I to	Coll	000	(22)	Ordinary
No. 14	I IDC Committee Australia 144	100	2007	(27)	Ordinary
5-IVIdy-14	UBS Securities Australia Ltd	le N	19,047	(2,100)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	7,800	(890)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	45	(2)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	22,104	(2,437)	Ordinary
3-May-14	UBS Securities Australia Ltd	Sell	16,462	(1,815)	Ordinary
08-May-14	I IDC CAN white A. whealing I and				
	UBS Securities Austrana Liu	Seil	089	(52)	Ordinary

08-May-14 08-May-14 08-May-14 08-May-14	UBS Securities Australia Ltd	Sell	1,362	(150)	Ordinary
1ay-14 1ay-14 1ay-14	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				(10.10)
1ay-14	UBS Securities Australia Ltd	Sell	109'1	(205)	Ordinary
1ay-14	UBS Securities Australia Ltd	Sell	2,833	(312)	Ordinary
44	UBS Securities Australia Ltd	Sell	2,561	(282)	Ordinary
UD-IVIAY-14	UBS Securities Australia Ltd	Sell	8,644	(952)	Ordinary
08-May-14	UBS Securities Australia Ltd	Seli	14,528	(1,600)	Ordinary
1dy-14	DDS Securities Australia Ltd	Set Con	43,539	(4,795)	Ordinary
00-May-14	UDC Counties Australia 14d	in S	4,524	(0/+)	Ordinary
08-May-14	I IRS Securities Australia (14)	3	ACT C	(00)	Ordinary
124-14	I IRS Secretias Australia I to	Jes	1 308	(000)	Ordinary
17-75	I IRS Securities A retrails 14d	Sall	005,	(10)	Ordinary
08-May-14	LIBS Securities Australia Ltd	Sall	252	(72)	Ordinary
De May 14	UDS Countries Australia 14d	100	000	(10)	Ordinary
10,14	Pri citato Accidina per cuo	le S	700 /	(6)	Ordinary
00-May-14	UDS Cocurities Australia Ltd	Call	100	(644)	Ordinary
37.14	USS Securities Australia Ltd	26	700,6	(000)	Ordinary
00-May-14	UDS Securities Australia Ltd	Call	505'S	(4204)	Ordinary
08-May-14	URS Secretifies Australia (14)	Ze II	750,24 507.5	(42),(1)	Ordinary
08-May-14	LIRS Securities Australia (14	Zell	36 597	(4 030)	Ordinary
08-May-14	I IRS Securities Australia I to	Zell	7697	(700)	Ordinary
08-May-14	UBS Securities Australia Itd	Sell	844	(65)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	2 107	(050)	Ordinary
08-Mav-14	UBS Securities Australia 1td	اع	672	(74)	Ordinary
08-May-14	UBS Securities Australia Ltd	llas	16	(101)	Ordinary
08-May-14	UBS Securities Australia I td	lleS	318	(35)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	3.314	(365)	Ordinary
08-May-14	LIBS Securities Australia Ltd	اع	6229	(686)	Ordinary
08-Mav-14	UBS Securities Australia Ltd	as	985	(59)	Ordinary
4av-14	UBS Securities Australia Ltd	Sell	163	(18)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	18.160	(2.000)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	1,907	(210)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	1,180	(130)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	1,308	(144)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	4,549	(201)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	16	(10)	Ordinary
08-Mav-14	UBS Securities Australia Ltd	Sell	281	(31)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	989	(JZ)	Ordinary
08-May-14	LIRS Sequenties Australia Ltd	Sell	791	(32)	Ordinary
00 May 14	TION CONTRACT Australia 142	Coll	1 053	(110)	Ordinary.
lay-14	Obs Securities Australia Liu	201	ccn'i	(10)	Ordinary
Ja-May-14	UBS Securities Australia Ltd	E C	5	(01)	Ordinary
1ay-14	UBS Securities Australia Ltd	Sell	2/5	(63)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	1,580	(174)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	3,732	(411)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	1,126	(124)	Ordinary
lav-14	UBS Securities Australia Ltd	Seil	16	(10)	Ordinary
1av-14	URS Securities Australia 11d	Spil	002	(60)	Ordinary
14-ve	I ISC Countries Australia 14	Cell	790.00	(0.50 6)	Ordinan
00 May 14	USC Countries Australia 14d	Coli	103,03	(400)	Ordinary
- (a)	ODS SECULINES AUSTRAINE CITY	120	2,032	(400)	Crainary
08-May-14	UBS Securities Australia Ltd	Sell	777	(57)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	2,134	(235)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	290	(65)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	7,264	(800)	Ordinary
lay-14	UBS Securities Australia Ltd	Sell	219	(89)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	7,101	(782)	Ordinary
08-Mav-14	UBS Securities Australia Ltd	8	23.481	(2.586)	Ordinary
08-May-14	UBS Securities Australia Ltd	95	4 386	(483)	Ordinary
08-May-14	I BS Securities Australia I to	3	3,886	(478)	Ordinary
08-May-14	I IRC Cocurities Australia I to	3 9	3 178	(350)	Ordinary
ay-14	Upo Commission Australia Ltd	100	201.0	(330)	Ordinary Ordinary
08-May-14	UBS Securities Australia Ltd	3	/0/1	(188)	Ordinary
08-May-14	UBS Securities Australia Ltd	- Se	18,560	(2,044)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	2,906	(320)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	18	(2)	Ordinary
	UBS Securities Australia Ltd	Seil	318	(32)	Ordinary
	IBS Securities Australia Ltd	Spill	182	(20)	Ordinary
1	Control of the second of the s	100	1 200	(24)	O Company

1998 1997 1998	08-May-14	DO Contribute Asset in Land	11.0	0000	(000)	100
USS Securities Australia Ltd Sell 1,5599 (1,718)	8-May-14	I IRS Sportifies Australia 14d	Jac.	3,300	(370)	Ordinary
Big Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Sell Securities Autorials Ltd Sell Sell Sell Sell Sell Securities Autorials Ltd Sell Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Securities Autorials Ltd Sell Sell Sell Securities Autorials Ltd Sell Sell Securities Aut	y-14	UBS Securities Australia Ltd	Sell	15,599	(1,718)	Ordinary
Big Securities Autorities Ltd Self 2.956 (320) (47	UBS Securities Australia Ltd	Sell	400	(44)	Ordinary
USS Securities Autorials Ltd Self Self Self Securities Autorials Ltd Self	1y-14	UBS Securities Australia Ltd	Iss	2 996	(330)	Ordinary
USS Securities Autorials Ltd Sell Sell Sell Securities Autorials Ltd Sell	1. 4.	UBS Securities Australia Ltd	<u></u>	2.751	(303)	Ordinary
USS Securities Autoriale Ltd Self Self Self Self Self Securities Autoriale Ltd Self	₹-14 14	UBS Securities Australia Ltd	Sell	30.936	(3.407)	Ordinary
UBS Securities Australia Ltd Sell Sell CS	y-14	UBS Securities Australia Ltd	las	345	(38)	Ordinary
UBS Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Sell Securities Australia tud Sell Sell Securities Australia tud Sell Sell Sell Sell Securities Australia tud Sell Sell Sell Securities Australia tud Sell S	¥-14	UBS Securities Australia Ltd	les	745	(82)	Ordinary
UBS Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Securities Australia tud Sell Sell Securities Australia tud Sell Sell Sell Securities Australia tud Sell S	¥-14	UBS Securities Australia Ltd	Sell	745	(82)	Ordinary
USS Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Sell C. 2624 C. 2639 C. 26	N-14	UBS Securities Australia 11d	3	2 971	(777)	Ordinary
UBS Securities Australia Ltd Sell 4,933 (540) UBS Securities Australia Ltd Sell 5,57 (581) UBS Securities Australia Ltd Sell 2,622 (289) UBS Securities Australia Ltd Sell 7,34 (589) UBS Securities Australia Ltd Sell 7,34 (693) UBS Securities Australia Ltd Sell 7,45 (623) UBS Securities Australia Ltd Sell 8,401 (632) UBS Securities Australia Ltd Sell 8,401 (632) UBS Securities Australia Ltd Sell 9,861 (1,108) UBS Securities Australia Ltd Sell 3,433 (1,086) UBS Securities Australia Ltd Sell 2,732 (39) UBS Securities Australia Ltd Sell 1,768 (1,108) UBS Securities Australia Ltd Sell 1,789 (1,287) UBS Securities Australia Ltd Sell 1,789 (1,287) UBS Securities Australia Ltd Sell 1,789 (1,287) UBS Securities Austra	V-14	UBS Securities Australia 11d	<u></u>	527	(58)	Control O
UBS Securities Australia Ltd Sell 5.27 (58) UBS Securities Australia Ltd Sell 2,624 (293) UBS Securities Australia Ltd Sell 2,624 (293) UBS Securities Australia Ltd Sell 8,881 (839) UBS Securities Australia Ltd Sell 7,155 (89) UBS Securities Australia Ltd Sell 7,155 (83) UBS Securities Australia Ltd Sell 7,155 (82) UBS Securities Australia Ltd Sell 7,155 (82) UBS Securities Australia Ltd Sell 7,155 (705) UBS Securities Australia Ltd Sell 7,156 (1,131) UBS Securities Australia Ltd Sell 7,156 (1,161) UBS Securities Australia Ltd Sell 7,266 (1,161) UBS Securities Australia Ltd Sell 7,266 (1,161) UBS Securities Australia Ltd Sell 7,266 (1,162) UBS Securities Australia Ltd Sell 7,266 (1,27) UBS Securities Australi	V-14	UBS Securities Australia Ltd	3	Z 903	(540)	Ordinary
(BS) Securities Australia Ltd Sell 2,662 (293) (BS) Securities Australia Ltd Sell 2,662 (289) (BS) Securities Australia Ltd Sell 7,25 (289) (BS) Securities Australia Ltd Sell 7,25 (89) (BS) Securities Australia Ltd Sell 7,135 (87) (BS) Securities Australia Ltd Sell 7,135 (1,086) (BS) Securities Australia Ltd Sell 7,689 (1,086) (BS) Securities Australia Ltd Sell 1,056 (1,108) (BS) Securities Australia Ltd Sell 1,056 (1,080) (BS) Securities Australia Ltd </td <td><u>₹</u></td> <td>UBS Securities Australia ! td</td> <td>3</td> <td>765</td> <td>(58)</td> <td>Coding</td>	<u>₹</u>	UBS Securities Australia ! td	3	765	(58)	Coding
UBS Securities Australia Ltd Sell 2,662 (293) UBS Securities Australia Ltd Sell 2,624 (289) UBS Securities Australia Ltd Sell 8,481 (934) UBS Securities Australia Ltd Sell 7,465 (623) UBS Securities Australia Ltd Sell 7,465 (623) UBS Securities Australia Ltd Sell 7,465 (757) UBS Securities Australia Ltd Sell 7,465 (757) UBS Securities Australia Ltd Sell 7,468 (822) UBS Securities Australia Ltd Sell 1,056 (1,168) UBS Securities Australia Ltd Sell 3,056 (1,204) UBS Securities Australia Ltd Sell 1,056 (1,169) UBS Securities Australia Ltd Sell 1,056 (1,206) UBS Securities	v 14	LiBS Securities Australia Ltd	3	7.2	(9)	Ordinary Ordinary
UBS Securities Australia Ltd Sell 2,624 (289) UBS Securities Australia Ltd Sell 745 (88) UBS Securities Australia Ltd Sell 745 (82) UBS Securities Australia Ltd Sell 2,665 (1,686) UBS Securities Australia Ltd Sell 8,465 (1,686) UBS Securities Australia Ltd Sell 8,405 (1,686) UBS Securities Australia Ltd Sell 7,468 (1,331) UBS Securities Australia Ltd Sell 1,056 (1,131) UBS Securities Australia Ltd Sell 1,056 (1,050) UBS Securities Australia Ltd Sell 1,056 (1,050) UBS Securities Australia Ltd Sell 1,056 (1,050) UBS Securi	V-14	UBS Securities Australia itd	3	7 667	(203)	Ordinary
UBS Securities Australia Ltd Sell 7.3 (8.2) UBS Securities Australia Ltd Sell 7.3 (8.2) UBS Securities Australia Ltd Sell 7.465 (7.364) UBS Securities Australia Ltd Sell 2.1465 (7.264) UBS Securities Australia Ltd Sell 7.135 (7.27) UBS Securities Australia Ltd Sell 7.465 (7.03) UBS Securities Australia Ltd Sell 7.466 (7.13) UBS Securities Australia Ltd Sell 7.466 (7.13) UBS Securities Australia Ltd Sell 7.666 (1.131) UBS Securities Australia Ltd Sell 7.666 (1.131) UBS Securities Australia Ltd Sell 7.766 (1.731) UBS Securities Australia Ltd Sell 7.766 (1.731) UBS Securities Australia Ltd Sell 7.726 (1.639) UBS Securities Australia Ltd Sell 7.736 (1.720) UBS Securities Australia Ltd Sell 7.736 (1.730) UBS Secur	14-14	UBS Securities Australia I td	3	2007	(080)	Ordinary Ordinary
UBS Securities Australia Ltd Sell 8,481 (934) UBS Securities Australia Ltd Sell 745 (82) UBS Securities Australia Ltd Sell 2,145 (1,086) UBS Securities Australia Ltd Sell 5,155 (1,086) UBS Securities Australia Ltd Sell 2,155 (1,057) UBS Securities Australia Ltd Sell 1,056 (1,161) UBS Securities Australia Ltd Sell 1,056 (1,162) UBS Securities Australia Ltd Sell 1,056 (1,163) UBS Securities Australia Ltd Sell 1,026 (1,163) UBS Securities Australia Ltd Sell 1,026 (1,203) UBS Securities Australia Ltd Sell 1,026 (1,203) UBS Securities Australia Ltd Sell 1,026 (1,203) UBS S	3v-14	UBS Securities Australia Ltd	- S	73	(8)	Ordinary
UBS Securities Australia Ltd Sell 7.45 (62) UBS Securities Australia Ltd Sell 21,465 (1,586) UBS Securities Australia Ltd Sell 9,81 (1,086) UBS Securities Australia Ltd Sell 405 (1,086) UBS Securities Australia Ltd Sell 4,05 (1,086) UBS Securities Australia Ltd Sell 1,066 (1,16) UBS Securities Australia Ltd Sell 1,026 (1,006) UBS Securities Australia Ltd Sell 1,026 (1,006) UBS Securities Australia Ltd Sell 1,006 (1,006) UBS Securiti	y-14	UBS Securities Australia Ltd		8.481	(934)	Ordinary
UBS Securities Australia Ltd Sell 21,465 (2,364) UBS Securities Australia Ltd Sell 5,861 (1,086) UBS Securities Australia Ltd Sell 472 (52) UBS Securities Australia Ltd Sell 6,405 (7,05) UBS Securities Australia Ltd Sell 1,056 (1,16) UBS Securities Australia Ltd Sell 1,056 (1,05) UBS Securities	37-14	UBS Securities Australia Ltd	Sel	745	(82)	Ordinary
UBS Securities Australia Ltd Sell 1,086 UBS Securities Australia Ltd Sell 6,405 UBS Securities Australia Ltd Sell 7,468 UBS Securities Australia Ltd Sell 1,056 UBS Securities Australia Ltd Sell 1,030 UBS Securities Australia Ltd <	3y-14	UBS Securities Australia Ltd	les.	21.465	(2.364)	Ordinary
UBS Securities Australia Ltd Sell 518 (57) UBS Securities Australia Ltd Sell 472 (52) UBS Securities Australia Ltd Sell 82,405 (705) UBS Securities Australia Ltd Sell 6,405 (705) UBS Securities Australia Ltd Sell 7,485 (822) UBS Securities Australia Ltd Sell 1,056 (1,131) UBS Securities Australia Ltd Sell 3,135 (1,151) UBS Securities Australia Ltd Sell 3,133 (1,162) UBS Securities Australia Ltd Sell 3,133 (1,162) UBS Securities Australia Ltd Sell 3,403 (1,162) UBS Securities Australia Ltd Sell 3,403 (1,62) UBS Securities Australia Ltd Sell 3,403 (1,65) UBS Securities Australia Ltd Sell 3,295 (1,65) UBS Securities Australia Ltd Sell 3,295 (1,65) UBS Securities Australia Ltd Sell 1,630 (1,65) UBS Securities Aus	94-14	UBS Securities Australia Ltd	Sell	9.861	(1.086)	Ordinary
Sell 472 (52) Sell 6,402 (52) Sell 6,403 (705) Sell 1,056 (116) Sell 3,431 (4,432) Sell 3,932 (374) Sell 3,932 (4,432) Sell 3,932 (4,432) Sell 3,932 (4,432) Sell 3,932 (4,432) Sell 1,034 (1,266) Sell 1,035 (1,266) Sell 1,032 (1,266)	38-May-14	UBS Securities Australia Ltd	lles	518	(57)	Ordinary
UBS Securities Australia Ltd Sell 82,401 (9,075) UBS Securities Australia Ltd Sell 6,405 (705) UBS Securities Australia Ltd Sell 7,468 (822) UBS Securities Australia Ltd Sell 1,026 (1,16) UBS Securities Australia Ltd Sell 1,056 (1,16) UBS Securities Australia Ltd Sell 1,056 (1,16) UBS Securities Australia Ltd Sell 3,433 (1,431) UBS Securities Australia Ltd Sell 3,433 (1,566) UBS Securities Australia Ltd Sell 3,433 (1,657) UBS Securities Australia Ltd Sell 3,995 (1,639) UBS Securities Australia Ltd Sell 1,092 (1,206) UBS Securities Australia Ltd Sell 1,092 (1,207) UBS	3y-14	UBS Securities Australia Ltd	Sell	472	(52)	Ordinary
UBS Securities Australia Ltd Sell 6,405 (705) UBS Securities Australia Ltd Sell 7,468 (235) UBS Securities Australia Ltd Sell 7,468 (8,21) UBS Securities Australia Ltd Sell 1,056 (1,131) UBS Securities Australia Ltd Sell 1,056 (1,131) UBS Securities Australia Ltd Sell 1,056 (1,16) UBS Securities Australia Ltd Sell 1,056 (1,69) UBS Securities Australia Ltd Sell 1,056 (1,69) UBS Securities Australia Ltd Sell 1,056 (1,69) UBS Securities Australia Ltd Sell 1,022 (1,204) UBS Securities Australia Ltd Sell 1,092 (1,05) UBS Securities Australia Ltd Sell 1,092 (1,05) UBS Securities Australia Ltd Sell 1,092 (1,00) UBS Securities Australia Ltd Sell 1,092 (1,00) UBS Securities Australia Ltd Sell 1,092 (1,00) UBS Securiti	34-14 14-14	UBS Securities Australia Ltd	Sell	82.401	(9.075)	Ordinary
UBS Securities Australia Ltd Sell 2,135 (253) UBS Securities Australia Ltd Sell 1,026 (1,15) UBS Securities Australia Ltd Sell 1,026 (1,16) UBS Securities Australia Ltd Sell 1,056 (1,16) UBS Securities Australia Ltd Sell 3,433 (1,206) UBS Securities Australia Ltd Sell 3,433 (1,206) UBS Securities Australia Ltd Sell 3,433 (1,206) UBS Securities Australia Ltd Sell 3,995 (1,206) UBS Securities Australia Ltd Sell 66,275 (1,206) UBS Securities Australia Ltd Sell 1,087 (1,206) UBS Securities Australia Ltd Sell 1,087 (1,207) UBS Securities Australia Ltd Sell 1,097 (1,097) <td< td=""><td>ay-14</td><td>UBS Securities Australia Ltd</td><td></td><td>6.405</td><td>(705)</td><td>Ordinary</td></td<>	ay-14	UBS Securities Australia Ltd		6.405	(705)	Ordinary
UBS Securities Australia Ltd Sell 7,468 (822) UBS Securities Australia Ltd Sell 1,0265 (1,131) UBS Securities Australia Ltd Sell 1,056 (1,16) UBS Securities Australia Ltd Sell 31,331 (374) UBS Securities Australia Ltd Sell 44,881 (4,932) UBS Securities Australia Ltd Sell 3,955 (1,16) UBS Securities Australia Ltd Sell 66,284 (7,284) UBS Securities Australia Ltd Sell 66,275 (1,390) UBS Securities Australia Ltd Sell 2,732 (1,391) UBS Securities Australia Ltd Sell 2,732 (1,295) UBS Securities Australia Ltd Sell 2,732 (1,295) UBS Securities Australia Ltd Sell 6,535 (1,795) UBS Securities Australia Ltd Sell 1,092 (1,795) UBS Securities Australia Ltd Sell 2,736 (450) UBS Securities Australia Ltd Sell 2,095 (1,796) U	34-14	UBS Securities Australia Ltd	Sell	2.135	(232)	Ordinary
UBS Securities Australia Ltd Sell 10,269 (1,131) UBS Securities Australia Ltd Sell 1,056 (1,131) UBS Securities Australia Ltd Sell 1,056 (1,16) UBS Securities Australia Ltd Sell 1,056 (1,16) UBS Securities Australia Ltd Sell 1,056 (1,290) UBS Securities Australia Ltd Sell 66,275 (1,204) UBS Securities Australia Ltd Sell 66,275 (1,204) UBS Securities Australia Ltd Sell 66,275 (1,204) UBS Securities Australia Ltd Sell 1,322 (1,005) UBS Securities Australia Ltd Sell 1,374 (1,027) UBS Securities Australia Ltd Sell 1,374 (1,027) UBS Securities Australia Ltd Sell 1,092 (1,027) UBS Securities Australia Ltd Sell 1,092 (1,007) UBS Securities Australia Ltd Sell 1,092 (1,007) UBS Securities Australia Ltd Sell 2,032 (1,007)	3V-14	UBS Securities Australia Ltd	les	7.468	(822)	Ordinary
UBS Securities Australia Ltd Seli 1,056 (116) UBS Securities Australia Ltd Seli 1,056 (1,16) UBS Securities Australia Ltd Seli 1,056 (1,16) UBS Securities Australia Ltd Seli 3,403 (1,302) UBS Securities Australia Ltd Seli 66,284 (7,283) UBS Securities Australia Ltd Seli 66,275 (1,306) UBS Securities Australia Ltd Seli 66,275 (7,283) UBS Securities Australia Ltd Seli 3,732 (360) UBS Securities Australia Ltd Seli 66,275 (7,283) UBS Securities Australia Ltd Seli 9,28 (1,050) UBS Securities Australia Ltd Seli 9,28 (1,050) UBS Securities Australia Ltd Seli 1,092 (1,200) UBS Securities Australia Ltd Seli 1,092 (1,200) UBS Securities Australia Ltd Seli 2,72 (300) UBS Securities Australia Ltd Seli 4,095 (450) UBS Sec	34-14	UBS Securities Australia Ltd	Sell	10.269	(1.131)	Ordinary
UBS Securities Australia Ltd Self 1,056 (116) UBS Securities Australia Ltd Self 31,331 (116) UBS Securities Australia Ltd Self 44,881 (4,923) UBS Securities Australia Ltd Self 44,881 (4,923) UBS Securities Australia Ltd Self 65,284 (7,284) UBS Securities Australia Ltd Self 66,284 (7,284) UBS Securities Australia Ltd Self 2,732 (439) UBS Securities Australia Ltd Self 2,732 (7284) UBS Securities Australia Ltd Self 2,732 (1595) UBS Securities Australia Ltd Self 9,88 (102) UBS Securities Australia Ltd Self 4,341 (475) UBS Securities Australia Ltd Self 4,341 (475) UBS Securities Australia Ltd Self 4,095 (450) UBS Securities Australia Ltd Self 2,726 (450) UBS Securities Australia Ltd Self 4,095 (450) UBS Securities	34-14	UBS Securities Australia Ltd	Sell	336	(37)	Ordinary
UBS Securities Australia Ltd Sell 1,056 (116) UBS Securities Australia Ltd Sell 3,433 (3,433) UBS Securities Australia Ltd Sell 4,488 (3,922) UBS Securities Australia Ltd Sell 1,7345 (1,906) UBS Securities Australia Ltd Sell 6,275 (1,906) UBS Securities Australia Ltd Sell 2,722 (390) UBS Securities Australia Ltd Sell 1,879 (1,655) UBS Securities Australia Ltd Sell 1,879 (1,655) UBS Securities Australia Ltd Sell 4,341 (1,735) UBS Securities Australia Ltd Sell 1,092 (1,02) UBS Securities Australia Ltd Sell 1,092 (1,02) UBS Securities Australia Ltd Sell 1,092 (1,005) UBS Securities Australia Ltd Sell 1,092 (1,005) UBS Securities Australia Ltd Sell 2,032 (450) UBS Securities Australia Ltd Sell 2,005 (450) UBS Sec	3y-14	UBS Securities Australia Ltd	Sell	1.056	(116)	Ordinary
UBS Securities Australia Ltd Sell 31,331 (3,443) UBS Securities Australia Ltd Sell 3,403 (374) UBS Securities Australia Ltd Sell 17,345 (1,906) UBS Securities Australia Ltd Sell 65,284 (7,284) UBS Securities Australia Ltd Sell 66,275 (7,283) UBS Securities Australia Ltd Sell 66,275 (7,283) UBS Securities Australia Ltd Sell 3,73 (350) UBS Securities Australia Ltd Sell 1,4,879 (1,635) UBS Securities Australia Ltd Sell 9,28 (1,02) UBS Securities Australia Ltd Sell 1,637 (1,29) UBS Securities Australia Ltd Sell 1,032 (1,20) UBS Securities Australia Ltd Sell 1,032 (1,20) UBS Securities Australia Ltd Sell 2,726 (8,20) UBS Securities Australia Ltd Sell 2,726 (8,20) UBS Securities Australia Ltd Sell 2,020 (1,40) UBS	3y-14	UBS Securities Australia Ltd	Self	1,056	(116)	Ordinary
UBS Securities Australia Ltd Sell 44,881 (4,922) UBS Securities Australia Ltd Sell 1,403 (374) UBS Securities Australia Ltd Sell 65,284 (7,284) UBS Securities Australia Ltd Sell 66,284 (7,284) UBS Securities Australia Ltd Sell 66,275 (7,283) UBS Securities Australia Ltd Sell 2,732 (300) UBS Securities Australia Ltd Sell 2,732 (35) UBS Securities Australia Ltd Sell 4,341 (172) UBS Securities Australia Ltd Sell 4,341 (477) UBS Securities Australia Ltd Sell 1,692 (1,65) UBS Securities Australia Ltd Sell 1,692 (450) UBS Securities Australia Ltd Sell 2,73 (1,49) UBS Securities Australia Ltd Sell 2,70 (1,45) UBS Securities Australia Ltd Sell 2,70 (1,45) UBS Securities Australia Ltd Sell 2,00 (1,45) UBS Securities A	3y-14	UBS Securities Australia Ltd	Sell	31,331	(3,443)	Ordinary
UBS Securities Australia Ltd Sell 3,403 (324) UBS Securities Australia Ltd Sell 6,284 (1,504) UBS Securities Australia Ltd Sell 6,275 (1,304) UBS Securities Australia Ltd Sell 3,995 (1,303) UBS Securities Australia Ltd Sell 2,722 (300) UBS Securities Australia Ltd Sell 1,879 (1,651) UBS Securities Australia Ltd Sell 4,341 (1,02) UBS Securities Australia Ltd Sell 4,341 (1,02) UBS Securities Australia Ltd Sell 4,341 (1,02) UBS Securities Australia Ltd Sell 1,022 (1,02) UBS Securities Australia Ltd Sell 1,022 (1,02) UBS Securities Australia Ltd Sell 1,022 (1,02) UBS Securities Australia Ltd Sell 2,736 (450) UBS Securities Australia Ltd Sell 2,035 (450) UBS Securities Australia Ltd Sell 2,036 (4,030) UBS Securiti	3y-14	UBS Securities Australia Ltd	Sell	14,881	(4,932)	Ordinary
UBS Securities Australia Ltd Sell 65,284 (1,906) UBS Securities Australia Ltd Sell 6,284 (7,284) UBS Securities Australia Ltd Sell 3,995 (4,393) UBS Securities Australia Ltd Sell 6,275 (7,283) UBS Securities Australia Ltd Sell 3,73 (350) UBS Securities Australia Ltd Sell 1,4,879 (1,635) UBS Securities Australia Ltd Sell 4,341 (4,77) UBS Securities Australia Ltd Sell 1,635 (1,20) UBS Securities Australia Ltd Sell 1,092 (1,20) UBS Securities Australia Ltd Sell 1,092 (1,20) UBS Securities Australia Ltd Sell 2,736 (450) UBS Securities Australia Ltd Sell 2,726 (849) UBS Securities Australia Ltd Sell 2,020 (1,400) UBS Securities Australia Ltd Sell 2,020 (1,600) UBS Securities Australia Ltd Sell 2,020 (1,600) UBS Se	ay-14	UBS Securities Australia Ltd	Sell	3,403	(374)	Ordinary
UBS Securities Australia Ltd Sell 66,284 (7,284) UBS Securities Australia Ltd Sell 66,277 (7,283) UBS Securities Australia Ltd Sell 2,722 (300) UBS Securities Australia Ltd Sell 2,722 (300) UBS Securities Australia Ltd Sell 9,28 (102) UBS Securities Australia Ltd Sell 4,341 (477) UBS Securities Australia Ltd Sell 4,341 (477) UBS Securities Australia Ltd Sell 1,092 (1,055) UBS Securities Australia Ltd Sell 1,092 (1,055) UBS Securities Australia Ltd Sell 1,092 (1,055) UBS Securities Australia Ltd Sell 2,03 (1,055) UBS Securities Australia Ltd Sell 2,09 (450) UBS Securities Australia Ltd Sell 2,09 (450) UBS Securities Australia Ltd Sell 2,09 (450) UBS Securities Australia Ltd Sell 2,09 (14) UBS Securities Austra	ly-14	UBS Securities Australia Ltd	Sell	17,345	(1,906)	Ordinary
UBS Securities Australia Ltd Sell 6,275 (439) UBS Securities Australia Ltd Sell 2,732 (300) UBS Securities Australia Ltd Sell 1,879 (1,635) UBS Securities Australia Ltd Sell 928 (102) UBS Securities Australia Ltd Sell 928 (102) UBS Securities Australia Ltd Sell 8,600 (947) UBS Securities Australia Ltd Sell 1,6335 (1,795) UBS Securities Australia Ltd Sell 1,6335 (1,795) UBS Securities Australia Ltd Sell 1,6335 (1,795) UBS Securities Australia Ltd Sell 1,092 (1,795) UBS Securities Australia Ltd Sell 2,735 (450) UBS Securities Australia Ltd Sell 2,095 (450) UBS Securities Australia Ltd Sell 2,095 (450) UBS Securities Australia Ltd Sell 1,095 (450) UBS Securities Australia Ltd Sell 1,095 (450) UBS Securities Aust	1y-14	UBS Securities Australia Ltd	Sell	66,284	(7,284)	Ordinary
UBS Securities Australia Ltd Sell 66,275 (7,283) UBS Securities Australia Ltd Sell 328 (300) UBS Securities Australia Ltd Sell 14,879 (1,639) UBS Securities Australia Ltd Sell 928 (1,037) UBS Securities Australia Ltd Sell 928 (102) UBS Securities Australia Ltd Sell 8,600 (945) UBS Securities Australia Ltd Sell 1,092 (1,20) UBS Securities Australia Ltd Sell 1,092 (1,20) UBS Securities Australia Ltd Sell 2,001 (450) UBS Securities Australia Ltd Sell 4,095 (450) UBS Securities Australia Ltd Sell 2,020 (450) UBS Securities Australia Ltd Sell 4,095 (450) UBS Securities Australia Ltd Sell 1,000 (1,000) UBS Securities Australia Ltd Sell 1,000 (1,000) UBS Securities Australia Ltd Sell 3,076 (1,60) UBS Securities Aust	1y-14	UBS Securities Australia Ltd	Sell	3,995	(439)	Ordinary
UBS Securities Australia Ltd Sell 2,722 (360) UBS Securities Australia Ltd Sell 328 (365) UBS Securities Australia Ltd Sell 537 (165) UBS Securities Australia Ltd Sell 4,341 (177) UBS Securities Australia Ltd Sell 4,341 (177) UBS Securities Australia Ltd Sell 1,092 (1,795) UBS Securities Australia Ltd Sell 1,092 (1,796) UBS Securities Australia Ltd Sell 1,092 (1,796) UBS Securities Australia Ltd Sell 2,736 (450) UBS Securities Australia Ltd Sell 2,096 (450) UBS Securities Australia Ltd Sell 2,096 (450) UBS Securities Australia Ltd Sell 2,096 (14) UBS Securities Australia Ltd Sell 3,076 (14) UBS Securities Australia Ltd Sell 3,076 (14) UBS Securities Australia Ltd Sell 3,076 (14) UBS Securities Australia Ltd </td <td>1y-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>66,275</td> <td>(7.283)</td> <td>Ordinary</td>	1y-14	UBS Securities Australia Ltd	Sell	66,275	(7.283)	Ordinary
UBS Securities Australia Ltd Sell 328 (36) UBS Securities Australia Ltd Sell 14,879 (1,635) UBS Securities Australia Ltd Sell 928 (102) UBS Securities Australia Ltd Sell 8,600 (945) UBS Securities Australia Ltd Sell 10,635 (1,795) UBS Securities Australia Ltd Sell 1,633 (1,795) UBS Securities Australia Ltd Sell 1,602 (1,796) UBS Securities Australia Ltd Sell 2,733 (1,796) UBS Securities Australia Ltd Sell 2,736 (450) UBS Securities Australia Ltd Sell 2,095 (450) UBS Securities Australia Ltd Sell 2,095 (450) UBS Securities Australia Ltd Sell 1,095 (450) UBS Securities Australia Ltd Sell 2,095 (450) UBS Securities Australia Ltd Sell 1,007 (1,000) UBS Securities Australia Ltd Sell 2,005 (2,000) UBS Securities Au	y-14	UBS Securities Australia Ltd	Sell	2,732	(300)	Ordinary
UBS Securities Australia Ltd Sell 14,879 (1,655) UBS Securities Australia Ltd Sell 928 (102) UBS Securities Australia Ltd Sell 4341 (477) UBS Securities Australia Ltd Sell 8,600 (945) UBS Securities Australia Ltd Sell 1,622 (120) UBS Securities Australia Ltd Sell 1,622 (120) UBS Securities Australia Ltd Sell 2,73 (30) UBS Securities Australia Ltd Sell 4,055 (450) UBS Securities Australia Ltd Sell 2,020 (450) UBS Securities Australia Ltd Sell 2,020 (30) UBS Securities Australia Ltd Sell 1,05 (1,00) UBS Securities Australia Ltd Sell 3,076 (60) UBS Securities Australia Ltd Sell 1,00 (1,000) UBS Securities Australia Ltd Sell 3,076 (1,000) UBS Securities Australia Ltd Sell 2,050 (1,000) UBS Securities Australia Lt	y-14	UBS Securities Australia Ltd	Sell	328	(36)	Ordinary
UBS Securities Australia Ltd Sell 928 (102) UBS Securities Australia Ltd Sell 4,341 (4702) UBS Securities Australia Ltd Sell 6,600 (945) UBS Securities Australia Ltd Sell 1,5335 (1,785) UBS Securities Australia Ltd Sell 1,6335 (1,785) UBS Securities Australia Ltd Sell 2,73 (30) UBS Securities Australia Ltd Sell 4,055 (450) UBS Securities Australia Ltd Sell 2,020 (222) UBS Securities Australia Ltd Sell 2,020 (222) UBS Securities Australia Ltd Sell 3,076 (450) UBS Securities Australia Ltd Sell 3,076 (16) UBS Securities Australia Ltd Sell 3,076 (16) UBS Securities Australia Ltd Sell 3,076 (16) UBS Securities Australia Ltd Sell 2,057 (2,200) UBS Securities Australia Ltd Sell 2,057 (2,200) UBS Securities Australia	ay-14	UBS Securities Australia Ltd	Sell	14,879	(1,635)	Ordinary
UBS Securities Australia Ltd Sell 928 (102) UBS Securities Australia Ltd Sell 4,341 (477) UBS Securities Australia Ltd Sell 16,335 (1,795) UBS Securities Australia Ltd Sell 1,092 (1,205) UBS Securities Australia Ltd Sell 2,73 (1,095) UBS Securities Australia Ltd Sell 2,73 (450) UBS Securities Australia Ltd Sell 4,095 (450) UBS Securities Australia Ltd Sell 2,095 (450) UBS Securities Australia Ltd Sell 2,095 (450) UBS Securities Australia Ltd Sell 1,07 (143) UBS Securities Australia Ltd Sell 2,095 (450) UBS Securities Australia Ltd Sell 1,07 (1,000) UBS Securities Australia Ltd Sell 3,076 (2,000) UBS Securities Australia Ltd Sell 1,00 (1,000) UBS Securities Australia Ltd Sell 2,00 (2,000) UBS Securities Austra	3y-14	UBS Securities Australia Ltd	Setl	537	(65)	Ordinary
UBS Securities Australia Ltd Sell 4,341 (477) UBS Securities Australia Ltd Sell 8,600 (945) UBS Securities Australia Ltd Sell 1,092 (1,205) UBS Securities Australia Ltd Sell 1,092 (1,205) UBS Securities Australia Ltd Sell 4,60 (1,055) UBS Securities Australia Ltd Sell 4,095 (450) UBS Securities Australia Ltd Sell 2,020 (222) UBS Securities Australia Ltd Sell 2,020 (222) UBS Securities Australia Ltd Sell 3,076 (16) UBS Securities Australia Ltd Sell 2,057 (2,26) UBS Securities Australia Ltd Sell 2,057 (2,28) UBS Securities Australia Ltd Sell 1,1019 (112) UBS Securities Australia	14-14	UBS Securities Australia Ltd	Sell	928	(102)	Ordinary
UBS Securities Australia Ltd Sell 6,600 (945) UBS Securities Australia Ltd Sell 1,092 (1,795) UBS Securities Australia Ltd Sell 1,092 (1,795) UBS Securities Australia Ltd Sell 273 (30) UBS Securities Australia Ltd Sell 4,055 (450) UBS Securities Australia Ltd Sell 7,726 (482) UBS Securities Australia Ltd Sell 7,726 (482) UBS Securities Australia Ltd Sell 2,020 (222) UBS Securities Australia Ltd Sell 3,076 (14) UBS Securities Australia Ltd Sell 3,076 (16) UBS Securities Australia Ltd Sell 3,076 (16) UBS Securities Australia Ltd Sell 3,076 (16) UBS Securities Australia Ltd Sell 2,057 (2,000) UBS Securities Australia Ltd Sell 2,057 (2,200) UBS Securities Australia Ltd Sell 2,057 (2,200) UBS Securities Australia	v-14	IUBS Securities Australia Ltd	Sell	4.341	(477)	Ordinary
UBS Securities Australia Ltd Sell 16,335 (1,755) UBS Securities Australia Ltd Sell 1,092 (120) UBS Securities Australia Ltd Sell 273 (120) UBS Securities Australia Ltd Sell 2,73 (30) UBS Securities Australia Ltd Sell 6,65 (4,55) UBS Securities Australia Ltd Sell 4,095 (4,65) UBS Securities Australia Ltd Sell 2,726 (849) UBS Securities Australia Ltd Sell 3,076 (1,600) UBS Securities Australia Ltd Sell 9,100 (1,000) UBS Securities Australia Ltd Sell 1,60 (1,000) UBS Securities Australia Ltd Sell 1,60 (1,000) UBS Securities Australia Ltd Sell 1,60 (1,000) UBS Securities Australia Ltd Sell 2,057 (2,500) UBS Securities Australia Ltd Sell 1,05 (1,000) UBS Securities Australia Ltd Sell 2,057 (2,200) UBS Securities A	v-14	UBS Securities Australia Ltd	la S	8,600	(945)	Ordinary
UBS Securities Australia Ltd Sell 1,092 (1/20) UBS Securities Australia Ltd Sell 164 (18) UBS Securities Australia Ltd Sell 46 (5) UBS Securities Australia Ltd Sell 4,095 (450) UBS Securities Australia Ltd Sell 7,726 (89) UBS Securities Australia Ltd Sell 2,020 (222) UBS Securities Australia Ltd Sell 3,076 (14) UBS Securities Australia Ltd Sell 9,100 (1,000) UBS Securities Australia Ltd Sell 9,100 (1,000) UBS Securities Australia Ltd Sell 1,166 (1,6) UBS Securities Australia Ltd Sell 1,166 (1,6) UBS Securities Australia Ltd Sell 1,207 (2,26) UBS Securities Australia Ltd Sell 1,109 (1,12) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd<	v-14	LIBS Securities Australia 11d	le D	16 335	(1 795)	Ordinary
UBS Scarrifies Australia Ltd Sell 162 UBS Scarrifies Australia Ltd Sell 273 (18) UBS Securifies Australia Ltd Sell 46 (5) UBS Securifies Australia Ltd Sell 4,055 (450) UBS Securifies Australia Ltd Sell 7,726 (849) UBS Securifies Australia Ltd Sell 2,020 (222) UBS Securifies Australia Ltd Sell 3,076 (14) UBS Securifies Australia Ltd Sell 9,100 (16) UBS Securifies Australia Ltd Sell 9,100 (16) UBS Securifies Australia Ltd Sell 1,166 (16) UBS Securifies Australia Ltd Sell 1,166 (16) UBS Securifies Australia Ltd Sell 2,057 (2,200) UBS Securifies Australia Ltd Sell 2,057 (2,200) UBS Securifies Australia Ltd Sell 1,019 (11,2) UBS Securifies Australia Ltd Sell 1,019 (11,2) UBS Securifies Australia Ltd Sell	v-14	LIRS Securities Australia I to		700.1	(00,1)	ordinal y
UBS Securities Australia Ltd Seli 273 (109) 109 (109) 10	14	LIRS Societies Australia 14d	No.	150,1	(150)	Ordinaly
UBS Securities Australia Ltd Sell 46 (5) 130	1	I IBS Securities Australia 14d	100	10.00	(00)	Ordinary
UBS Securities Australia Ltd Sell 40 Securities Australia Ltd 56 II (1055) UBS Securities Australia Ltd Sell 4,095 (450) (450) UBS Securities Australia Ltd Sell 2,020 (222) UBS Securities Australia Ltd Sell 3,076 (14) UBS Securities Australia Ltd Sell 9,100 (100) UBS Securities Australia Ltd Sell 9,100 (100) UBS Securities Australia Ltd Sell 146 (16) UBS Securities Australia Ltd Sell 160 (100) UBS Securities Australia Ltd Sell 2,057 (2,260) UBS Securities Australia Ltd Sell 18,200 (2,200) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,019 (1,12)	11/	I IDC Committee Augustia Ind	1150	2/2	(36)	Cidicialy
USS Securities Australia Ltd Sell 4,055 (450) USS Securities Australia Ltd Sell 4,055 (450) USS Securities Australia Ltd Sell 2,020 (322) USS Securities Australia Ltd Sell 3,076 (489) USS Securities Australia Ltd Sell 9,100 (1,000) USS Securities Australia Ltd Sell 9,100 (1,000) USS Securities Australia Ltd Sell 146 (1,000) USS Securities Australia Ltd Sell 18,200 (2,000) USS Securities Australia Ltd Sell 18,200 (2,000) USS Securities Australia Ltd Sell 18,200 (2,000) USS Securities Australia Ltd Sell 1,019 (1,12) USS Securities Australia Ltd Sell 1,019 (1,12) USS Securities Australia Ltd Sell 1,019 (1,12)	,	LIDO Constitute Australia 144	= 3	1090	(10.6)	y deliging
UBS Securities Australia Ltd Sell 7,725 (449) UBS Securities Australia Ltd Sell 7,026 (849) UBS Securities Australia Ltd Sell 2,020 (222) UBS Securities Australia Ltd Sell 3,076 (338) UBS Securities Australia Ltd Sell 9,100 (1,000) UBS Securities Australia Ltd Sell 146 (16) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 2,057 (2,260) UBS Securities Australia Ltd Sell 1,019 (1,10) UBS Securities Australia Ltd Sell 1,019 (1,10) UBS Securities Australia Ltd Sell 1,019 (1,10) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,019 (1,12)	1,	S IDC Consisting Australia 143	nac.	9,001	(1,055)	Ordinary
UBS Securities Australia Ltd Sell 7,720 (322) UBS Securities Australia Ltd Sell 127 (14) UBS Securities Australia Ltd Sell 3,076 (138) UBS Securities Australia Ltd Sell 9,100 (1,000) UBS Securities Australia Ltd Sell 146 (16) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,019 (1,12)		100 Counting Australia Ltd	100	255.5	(0,0)	Couring
UBS Securities Australia Ltd Sell 1.02 (14.4) UBS Securities Australia Ltd Sell 3.076 (1.38) UBS Securities Australia Ltd Sell 9,100 (1.000) UBS Securities Australia Ltd Sell 1.46 (1.60 UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,019 (1,12)	7 14	Libe Countition Australia 144	Sell Coll	07/1/	(049)	Ordinary
UBS Securities Australia Ltd Sell 3,076 (1,000) UBS Securities Australia Ltd Sell 9,100 (1,000) UBS Securities Australia Ltd Sell 9,100 (1,000) UBS Securities Australia Ltd Sell 5,400 (6,00) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 2,057 (226) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,019 (1,102) UBS Securities Australia Ltd Sell 1,165 (1,128) UBS Securities Australia Ltd 1,165 (1,128) UBS Securities Australia Ltd 1,165 (1,128) UBS Securities Australia Ltd 1,165 (1,128) UBS Securities Australia Ltd 1,165 (1,128) UBS Securities Australia Ltd 1,165 (1,128) UBS Securities Australia Ltd 1,165 (1,128) UBS Securities Australia Ltd 1,165 (1,128) UBS Secu	11/	LISS Securities Australia Ltd	Jac.	137	(14)	Ordinary
UBS Securities Australia Ltd Sell 9,100 (1,000) UBS Securities Australia Ltd Sell 146 (16) UBS Securities Australia Ltd Sell 146 (16) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 1,019 (1,102) UBS Securities Australia Ltd Sell 1,165 (1,12) UBS Securities Australia Ltd Sell 1,165 (1,288)	1	Libe countries Australia Ltd	100	2000	(200)	Cidirialy
UBS Securities Australia Ltd Sell 3.71 co. UBS Securities Australia Ltd Sell 1.46 (1.6) UBS Securities Australia Ltd Sell 18.200 (2.000) UBS Securities Australia Ltd Sell 18.200 (2.000) UBS Securities Australia Ltd Sell 18.200 (2.000) UBS Securities Australia Ltd Sell 1.019 (1.12) UBS Securities Australia Ltd Sell 1.165 (1.286)		LIDS Securities Australia Ltd	Sell	3,0/6	(338)	Ordinary
USS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 2,057 (226) UBS Securities Australia Ltd Sell 1,019 (1,20) UBS Securities Australia Ltd Sell 1,019 (1,20) UBS Securities Australia Ltd Sell 1,019 (1,20) UBS Securities Australia Ltd Sell 1,165 (1,28)	1	LIDS Socialities Australia Ltd	liac liac	3,100	(1,000)	Ordinary
UBS Securities Australia Ltd Sell 1,019 UBS Securities Australia Ltd Sell 2,057 (2,200) UBS Securities Australia Ltd Sell 2,057 (2,26) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,169 (1,12) UBS Securities Australia Ltd Sell 1,165 (1,28)		LIDE Commission August and	nec o	140	(01)	Ordinary
USS Securities Australia Ltd Sell 1,8,200 UBS Securities Australia Ltd Sell 2,057 (2,266) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,169 (1,12) UBS Securities Australia Ltd Sell 1,165 (1,288)	4	UBS Securities Australia Ltd	=	546	(90)	Ordinary
UBS Securities Australia Ltd Sell 2,057 (226) UBS Securities Australia Ltd Sell 18,200 (2,000) UBS Securities Australia Ltd Sell 1,019 (112) UBS Securities Australia Ltd Sell 1,165 (128)	y-14	UBS Securities Australia Ltd	Sell	18,200	(2,000)	Ordinary
UBS Securities Australia Ltd Sell 1,8200 (2,000) UBS Securities Australia Ltd Sell 1,019 (1,12) UBS Securities Australia Ltd Sell 1,165 (1,28)	4 14	UBS Securities Australia Ltd	Sell	2,057	(526)	Ordinary
UBS Securities Australia Ltd Sell 1,019 (112)	4:	UBS Securities Australia Ltd	Seil	18,200	(2,000)	Ordinary
UBS Securities Australia (128)	y-14	UBS Securities Australia Ltd	Sell	1,019	(112)	Ordinary
(07)	717	71 20 1 2 1 1 1 1				- Commercial Commercia

Sell
Sell
Sell
Sell
Sell
Sell
Sell
Sell
ll dy

08-May-14 08-May-14	Obs securities Australia Ltu		4,683	(5.4)	Ordinary
8-May-14	MY Sectional Aristralia 12	=97	7 0 2	(090)	
	UBS Securities Australia Ltd	- S	1 840	(202)	Ordinary
08-Mav-14	UBS Securities Australia Ltd	3	18 921	(7707)	Ordinary
D8-May-14	LIBS Securities Australia Ltd	3	301	(20)	Ordinary
08-May-14	UBS Securities Australia Ltd	9	182	000	Ordinary
38-May-14	UBS Securities Australia Ltd		64	(2)	Ordinary
3-May-14	UBS Securities Australia Ltd		501	(52)	Ordinary
08-May-14	UBS Securities Australia Ltd	lles.	1.321	(145)	Ordinary
08-May-14	UBS Securities Australia Ltd	les	17.528	(1.924)	Ordinary
08-May-14	UBS Securities Australia Ltd	les	2 134	(734)	Ordinary
08-May-14	UBS Securities Australia Ltd	las.	19 104	(7 097)	Ordinary
08-May-14	UBS Securities Australia Ltd	les	10.540	(1,157)	Ordinary
08-May-14	UBS Securities Australia Ltd	las	5 879	(646)	Ordinary
08-May-14	UBS Securities Australia Ltd	الع	582	(64)	Spinal
38-May-14	UBS Securities Australia Itd	lay	13 941	(1 530)	Ordinary.
08-May-14	LIBS Securities Australia I td	100	3.367	(255,1)	Ordinary
08-May-14	I IRS Securities Australia 14d	le S	195'5	(0/5)	Ordinary
18-May-17	I IBC Counties Australia 14d	100	020'1	(0/1)	Ordinary
18-May-14	Libo Counting Australia 144	lan d	0,554	(656)	Ordinary
OS-May 14	TIDS Consistes Australia Ltd	120	5/1/2	(828)	Ordinary
Paridy 14	Upp Securities Australia Ltd	III.	7,392	(153)	Ordinary
John Jan	UBS Securities Australia Ltd	- X	610,1	(112)	Ordinary
Jo-May-14	UBS Securities Australia Ltd	Sell	24,943	(2,741)	Ordinary
JS-May-14	UBS Securities Australia Ltd	Sell	2,730	(300)	Ordinary
38-May-14	UBS Securities Australia Ltd	Self	410	(45)	Ordinary
3-May-14	UBS Securities Australia Ltd	Sell	637	(02)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	16,426	(1,805)	Ordinary
08-May-14	UBS Securities Australia Ltd	Seil	930	(102)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	5.575	(612)	Ordinary
8-May-14	UBS Securities Australia Itd	اق	592	(510)	Color of the Color
May-14	I IBS Securities Australia (14)	lles	2000	(000)	000
00 May 14	U.O.C. Constitutes Australia Ltd	liac	011/6	(000)	Ordinary
20-14lay-14	ODS Securities Australia Liu	Jac I	3,002	(402)	Ordinary
-Ividy-14	UDS SECURITIES AUSTRAIRA LTG	Self	778,1	(200)	Ordinary
J8-May-14	UBS Securities Australia Ltd	Sell	3,535	(388)	Ordinary
J8-May-14	Securities Austra	Seli	3,152	(346)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	3,161	(347)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	2,733	(300)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	2.605	(286)	Ordinary
8-May-14	UBS Securities Australia I td	9	10.094	(1 108)	Ordinan
18-May-14	I IRS Secretion Australia 11d		10,031	1000/	Cidilary
00 Mov 14	TIDE Commission Australia Lau	11-01	1/6/6	(0/6)	Ordinary
May 14	UBS Securities Australia Ltd	Seil	4,555	(200)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	711	(28)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	3,571	(392)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	8.235	(904)	Ordinary
8-May-14	ILIBS Sequitibes Australia Ltd	9	76	60	i de la la la la la la la la la la la la la
AL MAN OF	1 DC Consister American 1+4	11-2	17	(C)	O'GILIBIY
-INIAy-14	Obs Securities Australia LIG	Seli	617'1	(134)	Ordinary
J8-May-14	UBS Securities Australia Ltd	Seli	19,565	(2,150)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	1,820	(200)	Ordinary
28-May-14	UBS Securities Australia Ltd	Sell	2,621	(588)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	18.136	(1,993)	Ordinary
38-May-14	UBS Securities Australia Ltd	le y	PU3 E	(385)) raining
18.May-14	I IBC Cocurities Australia 144		25.00	(00)	O'CITION O
	The Contract Australia Lite	liec i	5/7	(30)	Ordinary
JO-IVIGY- 14	Obs securities Australia Lid)eii	12,504	(1,385)	Ordinary
US-May-14	UBS Securities Australia Ltd	Sell	3,367	(370)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	1,847	(203)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	9,446	(1,038)	Ordinary
08-May-14	UBS Securities Australia Ltd	ll se	946	(104)	Ordinary
-Mav-14	LIRS Securities Australia Ltd	3	000	(156)	Ordina
OR-May-14	DO Counting Australia	100	224,	(00)	S C C C C
May 14	ODS Securities Australia Liu	i o	986,51	(1,493)	Ordinary
US-May-14	UBS Securities Australia Ltd	Seil	3,549	(390)	Ordinary
J8-May-14	UBS Securities Australia Ltd	Sell	6	(1)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	729	(08)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	o	(1)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	103 C	(300)	, accipio
1,727.1					VIETE I
- 47	I IDC Congrition Arietralia I to	= 0	400,0	(1000)	Crainary

I IDC Commission Americal and	11-31	,,,,,	400	. (
UBS Securities Australia Ltd	Sell	3,221	(354)	Ordinary
3S Securities Australia Ltd	Sell	8.791	(996)	Ordinary
UBS Securities Australia Ltd	Sell	865	(96)	Ordinary
35 Securities Australia Ltd	Sell	68,250	(7,500)	Ordinary
UBS Securities Australia Ltd	Sell	473	(52)	Ordinary
JBS Securities Australia Ltd	Sell	291	(32)	Ordinary
JBS Securities Australia Ltd	Sell	3.158	(347)	Ordinary
35 Securities Australia Ltd	les	501	(55)	Ordinary
UBS Securities Australia Ltd	Sell	7.380	(811)	Ordinary
BS Securities Australia Ltd	lles	455	(50)	Ordinary
JBS Securities Australia Ltd	Ses	20.712	(2, 276)	Ordinary
JBS Securities Australia Ltd	Sell	16	(10)	Ordinary
JBS Securities Australia Ltd	Sell	91	(01)	Ordinary
BS Securities Australia I to	9	6 207	(603)	, and a
IBS Securities Australia Itd	S	404	(550)	Conditions
IBC Contrition Australia Ltd	llac llac	010	(54)	Ordinary
ios seculides Australia Lio	ne.	218	(24)	Ordinary
JBS securities Australia Ltd	Sell	373	(41)	Ordinary
UBS Securities Australia Ltd	Sell	100	(11)	Ordinary
BS Securities Australia Ltd	Sell	127	(14)	Ordinary
BS Securities Australia Ltd	Sell	2,139	(235)	Ordinary
UBS Securities Australia Ltd	Sell	33,561	(3,688)	Ordinary
JBS Securities Australia Ltd	lles	91	(10)	Ordinary
IBS Securities Australia I to	9	6161	(223)	Ordina
IBS Securities Australia I to	Sell	7,07	(090)	Colorary
RS Securities Australia 14d	100	010	(101)	Ordinary
IRC Committee Americal and	300	010 0	(101)	Cidinary
IBC Consistion Australia Ind	liac liac	040,0	(400)	Orginary
D3 Securities Australia Liu	II-SC	510,55	(3,034)	Ordinary
UBS Securities Australia Ltd	Sell	29,930	(3,289)	Ordinary
BS Securities Australia Ltd	Sell	2,457	(270)	Ordinary
BS Securities Australia Ltd	Sell	7,990	(878)	Ordinary
BS Securities Australia Ltd	Seil	3,476	(382)	Ordinary
UBS Securities Australia Ltd	Sell	4,359	(479)	Ordinary
UBS Securities Australia Ltd	Sell	200	(22)	Ordinary
UBS Securities Australia Ltd	Sell	896'8	(436)	Ordinary
UBS Securities Australia Ltd	Sell	146	(91)	Ordinary
UBS Securities Australia Ltd	Sell	455	(20)	Ordinary
BS Securities Australia Ltd	Sell	291	(32)	Ordinary
UBS Securities Australia ! td	la y	883	(42)	Ordinary
IRS Societies Australia 114	500	7 580	(1/0)	O Citiliary
Do contines Australia tid	Dell Dell	600,7	(834)	Ordinary
bs securities Australia Ltd	Sell	4,(3)	(454)	Ordinary
UBS Securities Australia Ltd	Sell	33,224	(3,651)	Ordinary
BS Securities Australia Ltd	Sell	483	(53)	Ordinary
UBS Securities Australia Ltd	Sell	1,484	(163)	Ordinary
JBS Securities Australia Ltd	Sell	14,058	(1,544)	Ordinary
JBS Securities Australia Ltd	105	4 077	(448)	Ordinary
JBS Securities Australia Ltd	la	364	(40)	Ordinan
IBS Securities Australia I to	100		(11)	Ordinary
BS Sequities Australia 14d	100	201	(44)	Ordinary
IDC Committees According Line	100	10/	(2)	Ordinary
IDS Commission Assessed and	120	100	(1)	Ordinary
BS Seculities Australia Lid	lisc list	/71	(4)	Ordinary
Jobs Securities Australia Ltd	Seil	7,132	(300)	Orginary
JBS Securities Australia Ltd	Sell	8,190	(006)	Ordinary
JBS Securities Australia Ltd	Sell	15,561	(1,710)	Ordinary
8S Securities Australia Ltd	Sell	3,005	(330)	Ordinary
JBS Securities Australia Ltd	Sell	500	(23)	Ordinary
JBS Securities Australia Ltd	Sell	1,128	(124)	Ordinary
UBS Securities Australia Ltd	les	2,730	(300)	Ordinary
3S Securities Australia Ltd	Sell	282	(31)	Ordinary
UBS Securities Australia Ltd	llas	673	(74)	Ordinary
IRS Securities Australia I to	3	100	(F.)	Ordinary
IDC Convision America 14d	100	192	(00)	Ordinary
of Committee Australia Ltd	liac I	150,55	(3,026)	Ordinary
Obsisted Australia Ltd	Jac J	6,523	(917)	Ordinary
UBS Securities Australia Ltd	Seil	1,440	(88)	Ordinary
DC Consistent Australia 140			/	Ciciliary
no permittee Anamaila Fin	Sell	14,785	(1,622)	Ordinary

	Do May 14	Libe Commission Assessed at the	===			7000
1975 Securities Australia Ltd 5481 1,2,26 1,0,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1	May-14	UBS Securities Australia Ltd	Sell	5,9/9	(65/)	200
Bits Securities Australia Ltd Self Securities Australia Lt	May-14	UBS Securities Australia Ltd		7 476	(816)	O Grinary
Securities Australia Ltd Self Self Securities Australia Ltd Self Se	May-14	UBS Securities Australia I td	J	5 542	(6/6)	Ordinary
Comparison of the comparison	May 14	I IBC Constitution Australia Ltd	====	245,0	(600)	S S S S S S S S S S S S S S S S S S S
1982 Securities Autoratials Ltd 5481 137,928 (15,157) (1	1000	Opp Securities Australia Ltd	llac llac	4,805	(979)	Ordinary
DES Securities Autoratials Ltd Sell 25.859 (15.1) DES Securities Autoratials Ltd Sell 25.879 (5.1) DES Securities Autoratials Ltd Sell 25.20 (2.67) DES Securities Autoratials Ltd Sell 38.273 (4.27) DES Securities Autoratials Ltd Sell 38.273 (4.27) DES Securities Autoratials Ltd Sell 38.273 (4.27) DES Securities Autoratials Ltd Sell 38.274 (4.27) DES Securities Autoratials Ltd Sell 1.800 (2.70)	May 14	Vide Consider Australia Liu		8,845	(3/5)	Ordinary
URS Securities Australia Ltd Sell 2,8,39 (2,8) URS Securities Australia Ltd Sell 2,8,39 (2,8) BIS Securities Australia Ltd Sell 2,8,39 (2,8) BIS Securities Australia Ltd Sell 2,8,39 (4,28) BIS Securities Australia Ltd Sell 3,653 (4,28) BIS Securities Australia Ltd Sell 3,653 (4,28) BIS Securities Australia Ltd Sell 1,800 (7,21) BIS Securities Australia Ltd Sell 1,800 (7,20) BIS Securiti	1viay-14	Obs Securities Australia Ltd	Jac.	676'/81	(15,15/)	Ordinary
UBS Securities Australia Ltd Sell 8,019 (8,91) BBS Securities Australia Ltd Sell 28,574 (4,280) BBS Securities Australia Ltd Sell 38,574 (4,280) BBS Securities Australia Ltd Sell 38,574 (4,280) BBS Securities Australia Ltd Sell 1,522 (20) BBS Securities Australia Ltd Sell 1,500 (20) BBS Securities Australia Ltd Sell 1,800 (20) BBS Securities Australia Ltd<	Nidy-14	UBS Securities Australia Ltd	Net.	658,42	(2,871)	Ordinary
UBS Securities Australia Ltd Sell 28,305 (3,45) BBS Securities Australia Ltd Sell 28,305 (3,45) LBS Securities Australia Ltd Sell 36,53 (407) LBS Securities Australia Ltd Sell 1,256 (2.14) LBS Securities Australia Ltd Sell 1,266 (2.14) LBS Securities Australia Ltd Sell 1,350 (2.14) LBS Securities Australia Ltd Sell 1,300 (2.20) LBS Securities Australia Ltd Sell 1,300 (2.70) LBS Securiti	May-14	UBS Securities Australia Ltd	Xeil	8,019	(891)	Ordinary
UBS Securities Australia Ltd Sell Sell Sell Sell Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Sell Sell Securities Australia Ltd Sell Sell Sell Securities Australia Ltd Sell Sell Sell Sell Sell Sell Securities Australia Ltd Sell Sel	May-14	UBS Securities Australia Ltd	Sell	28,305	(3,145)	Ordinary
UBS Securities Australia Ltd Sell Sell Sept	May 14	UBS Securities Australia Ltd	Sell	459	(51)	Ordinary
UBS Securities Australia Ltd Sell Sell 36,574 (4,07) UBS Securities Australia Ltd Sell 36,07 36,07 UBS Securities Australia Ltd Sell 5,01 1,000 UBS Securities Australia Ltd 5,01 5,01 1,000 UBS Securities Australia Ltd 5,01 5,01 1,000 UBS Securities Australia Ltd 5,01 5,01 1,000	-May-14	UBS Securities Australia Ltd	Sell	738	(82)	Ordinary
UBS Securities Matrialia Ltd Sell 3,663 (407) UBS Securities Matrialia Ltd Sell 1,256 (214) UBS Securities Matrialia Ltd Sell 1,303 (27) UBS Securities Matrialia Ltd Sell 1,800 (15) UBS Securities Matrialia Ltd Sell 1,800 (27) UBS Securities Matrialia Ltd Sell 1,800 (27) UBS Securities Matrialia Ltd Sell 1,800 (27) UBS Securities Matrialia Ltd Sell 1,800 (20) UBS Securities Matrialia Ltd	May-14	UBS Securities Australia Ltd	Seli	38,574	(4,286)	Ordinary
UBS Securities Materials Ltd Sell 1,552 (224) UBS Securities Materials Ltd Sell 1,500 (57) UBS Securities Materials Ltd Sell 1,800 (50) UBS Securities Materials Ltd Sell 1,800 (50) UBS Securities Materials Ltd Sell 1,800 (20) UBS Securities Materials Ltd	May-14	UBS Securities Australia Ltd	Seli	3,663	(407)	Ordinary
UBS Securities Australia Ltd Sell 1,926 (21) UBS Securities Australia Ltd Sell 1,800 (20) UBS Securities Australia Ltd Sell 1,800 (20) UBS Securities Australia Ltd Sell 2,43 (27) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Seli	252	(28)	Ordinary
UBS Securities Australia Ltd Sell \$15 (57) UBS Securities Australia Ltd Sell 1300 (70) UBS Securities Australia Ltd Sell 387 (43) UBS Securities Australia Ltd Sell 222 (28) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	1,926	(214)	Ordinary
USS Securities Australia Ltd Sell 1 800 (200) UBS Securities Australia Ltd Sell 135 (43) UBS Securities Australia Ltd Sell 243 (43) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 441 (49) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 5,702 (30) UBS Securities Australia Ltd Sell 5,80 (30) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	513	(57)	Ordinary
UBS Securities Australia Ltd Sell 135 (15) UBS Securities Australia Ltd Sell 227 (20) UBS Securities Australia Ltd Sell 227 (20) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Sell	1,800	(200)	Ordinary
UBS Securities Australia Ltd Sell 387 (43) UBS Securities Australia Ltd Sell 1,800 (20) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 657 (49) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	135	(15)	Ordinary
UBS Securities Australia Ltd Sell 243 (27) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 644 (72) UBS Securities Australia Ltd Sell 643 (73) UBS Securities Australia Ltd Sell 657 (80) UBS Securities Australia Ltd Sell 657 (80) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 2,702 (30) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd <t< td=""><td>May-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>387</td><td>(43)</td><td>Ordinary</td></t<>	May-14	UBS Securities Australia Ltd	Sell	387	(43)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 648 (72) UBS Securities Australia Ltd Sell 657 (73) UBS Securities Australia Ltd Sell 441 (49) UBS Securities Australia Ltd Sell 441 (49) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,900 (200) UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Seil	243	(22)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 648 (72) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 657 (73) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	las.	252	(28)	Ordina
UBS Securities Australia Ltd Sell 1,800 (200 UBS Securities Australia Ltd Sell 1,800 (200 UBS Securities Australia Ltd Sell 1,800 (200 UBS Securities Australia Ltd Sell 657 (73) UBS Securities Australia Ltd Sell 441 (49) UBS Securities Australia Ltd Sell 441 (49) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd Sell 1,900 (201) UBS Securities Australia Ltd Sell 2,702 (30) UBS Securities Australia Ltd Sell 2,702 (30) UBS Securities Australia Ltd Sell 2,702 (30) UBS Securities Australia Ltd <	May-14	LIRS Securities Australia Ltd	9	008 \$	(002)	Ordinary
UBS Securities Australia Ltd Sell 1,500 (127) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 657 (73) UBS Securities Australia Ltd Sell 657 (73) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd	May-14	I IRS Securities Australia I to	100	000	(002)	Columbia
UBS Securities Australia Ltd Sell 15.00 (2.00) UBS Securities Australia Ltd Sell 5.31 (57) UBS Securities Australia Ltd Sell 65.7 (73) UBS Securities Australia Ltd Sell 44.1 (49) UBS Securities Australia Ltd Sell 45.9 (50) UBS Securities Australia Ltd Sell 1.800 (200) UBS Securities Australia Ltd Sell 1.800 (201) UBS Securities Australia Ltd Sell 2.702 (30) UBS Securities Australia Ltd Sell 2.702 (30) UBS Securities Australia Ltd Sell 1.700 (11) UBS Securities Australia Ltd Sell 1.700 (12) UBS Securities Australia Ltd	May 14	I IBC Contribut Australia 144		000/1	(007)	Collinary
UBS Securities Australia Ltd Sell 657 (73) UBS Securities Australia Ltd Sell 657 (73) UBS Securities Australia Ltd Sell 441 (49) UBS Securities Australia Ltd Sell 459 (51) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,909 (720) UBS Securities Australia Ltd	May-14	1 IBC Contrition Australia 14d	lisa.	0100	(7/)	Ordinary
UBS Securities Australia Ltd Sell 657 (73) UBS Securities Australia Ltd Sell 447 (49) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,702 (30) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd	1	De Contract A retain Co.	100	006,1	(50)	Cidinary
UBS Securities Australia Ltd Sell 44.1 (49) UBS Securities Australia Ltd Sell 45.9 (51) UBS Securities Australia Ltd Sell 45.9 (51) UBS Securities Australia Ltd Sell 1.800 (200) UBS Securities Australia Ltd Sell 1.800 (201) UBS Securities Australia Ltd Sell 1.800 (201) UBS Securities Australia Ltd Sell 2.702 (30) UBS Securities Australia Ltd Sell 2.702 (30) UBS Securities Australia Ltd Sell 2.702 (30) UBS Securities Australia Ltd Sell 1.72 (30) UBS Securities Australia Ltd Sell 1.70 (30) UBS Securities Australia Ltd Sell 1.909 (21) UBS Securities Australia Ltd	V ay 14	Cos seculities Australia Liu	100	513	(/4)	Ordinary
UBS Securities Australia Ltd Sell 441 (49) UBS Securities Australia Ltd Sell 588 (62) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 9,91 (11) UBS Securities Australia Ltd Sell 801 (11) UBS Securities Australia Ltd Sell 8,40 (11) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,909 (200) UBS Securities Australia Ltd Sell 1,909 (200) UBS Securities Australia Ltd	May-14	Ups securities Australia Ltd	Sell	/59	(/3)	Ordinary
UBS Securities Australia Ltd Sell 458 (62) UBS Securities Australia Ltd Sell 459 (50) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 80 (11) UBS Securities Australia Ltd Sell 2,702 (30) UBS Securities Australia Ltd Sell 2,702 (30) UBS Securities Australia Ltd Sell 1,702 (30) UBS Securities Australia Ltd Sell 1,800 (2,12) UBS Securities Australia Ltd <	May-14	UBS Securities Australia Ltd	Sell	441	(49)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (510) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 801 (89) UBS Securities Australia Ltd Sell 801 (89) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,008 (112) UBS Securities Australia Ltd Sell 1,008 (112) UBS Securities Australia Ltd Sell 1,008 (100) UBS Securities Australia Ltd Sell 1,008 (100) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	558	(62)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 342 (74) UBS Securities Australia Ltd Sell 99 (11) UBS Securities Australia Ltd Sell 801 (81) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,909 (1,2) UBS Securities Australia Ltd Sell 1,000 (200) UBS Securities Australia Ltd Sell 1,000 (200) UBS Securities Australia Ltd Sell 1,000 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	459	(51)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd Sell 801 (89) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (201) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd <td>-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>1,800</td> <td>(200)</td> <td>Ordinary</td>	-May-14	UBS Securities Australia Ltd	Sell	1,800	(200)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (20) UBS Securities Australia Ltd Sell 1,800 (20) UBS Securities Australia Ltd Sell 656 (74) UBS Securities Australia Ltd Sell 99 (11) UBS Securities Australia Ltd Sell 801 (89) UBS Securities Australia Ltd Sell 3,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	1,800	(200)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 80 (11) UBS Securities Australia Ltd Sell 80 (11) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,900 (212) UBS Securities Australia Ltd Sell 1,000 (212) UBS Securities Australia Ltd Sell 1,000 (212) UBS Securities Australia Ltd Sell 1,000 (212) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	234	(50)	Ordinary
UBS Securities Australia Ltd Sell 3.22 (38) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 99 (11) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	1,800	(200)	Ordinary
UBS Securities Australia Ltd Sell 1560 (74) UBS Securities Australia Ltd Sell 1800 (200) UBS Securities Australia Ltd Sell 801 (891) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 792 (88) UBS Securities Australia Ltd Sell 792 (80) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,909 (720) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Sell	342	(38)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 80 (11) UBS Securities Australia Ltd Sell 80 (11) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 782 (88) UBS Securities Australia Ltd Sell 792 (300) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,00 (1,12) UBS Securities Australia Ltd Sell 1,909 (2,12) UBS Securities Australia Ltd Sell 1,800 (2,00) UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Jies	999	(74)	Ordinary
UBS Securities Australia Ltd Sell 99 (11) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,70 (13) UBS Securities Australia Ltd Sell 1,70 (10) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (102) UBS Securities Australia Ltd Sell 1,800 (102) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	1,800	(200)	Ordinary
UBS Securities Australia Ltd Sell 807 (89) UBS Securities Australia Ltd Sell 2.702 (300) UBS Securities Australia Ltd Sell 2.702 (300) UBS Securities Australia Ltd Sell 2.702 (300) UBS Securities Australia Ltd Sell 1.800 (200) UBS Securities Australia Ltd Sell 7.702 (300) UBS Securities Australia Ltd Sell 1.702 (300) UBS Securities Australia Ltd Sell 1.008 (112) UBS Securities Australia Ltd Sell 1.909 (212) UBS Securities Australia Ltd Sell 1.800 (200) UBS Securities Australia Ltd Sell 8.108 (102) UBS Securities Australia Ltd Sell 8.108 (145) UBS Securities Australia Ltd <td>Mav-14</td> <td>UBS Securities Australia Ltd</td> <td>195</td> <td>000</td> <td>(11)</td> <td>Ordinary</td>	Mav-14	UBS Securities Australia Ltd	195	000	(11)	Ordinary
UBS Securities Australia Ltd Sell 2,702 (30) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,902 (89) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 504 (56) UBS Securities Australia Ltd Sell 504 (45) UBS Securities Australia Ltd Sell 504 (45) UBS Securities Australia Ltd Sell 8,109 (145) UBS Securities Australia Ltd	May-14	II IRS Sequenties Australia Ltd	-	100	(00)	Ordinary
UBS Securities Australia Ltd Sell 2,702 (50) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 792 (38) UBS Securities Australia Ltd Sell 772 (30) UBS Securities Australia Ltd Sell 1,702 (112) UBS Securities Australia Ltd Sell 1,909 (2,12) UBS Securities Australia Ltd Sell 1,909 (2,12) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (40) UBS Securities Australia Ltd Sell 1,800 (40) UBS Securities Australia Ltd	May 14	I IDC Convision Australia Ltd	nac u	108	(600)	Oldinary
UBS Securities Australia Ltd Sell 2 702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 792 (89) UBS Securities Australia Ltd Sell 1,702 (130) UBS Securities Australia Ltd Sell 1,009 (2,12) UBS Securities Australia Ltd Sell 1,909 (2,12) UBS Securities Australia Ltd Sell 1,800 (2,00) UBS Securities Australia Ltd Sell 8,109 (102) UBS Securities Australia Ltd Sell 8,109 (145) UBS Securities Australia Ltd Sell 8,60 (145) UBS Securities Australia	Widy-14	Obs securities Australia Lid	Sell	70,107	(300)	Ordinary
UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 722 (88) UBS Securities Australia Ltd Sell 1,77 (1,3) UBS Securities Australia Ltd Sell 1,008 (1,12) UBS Securities Australia Ltd Sell 1,008 (1,12) UBS Securities Australia Ltd Sell 1,800 (2,12) UBS Securities Australia Ltd Sell 1,800 (2,00) UBS Securities Australia Ltd Sell 8,109 (9,1) UBS Securities Australia Ltd Sell 8,109 (102) UBS Securities Australia Ltd Sell 8,109 (102) UBS Securities Australia Ltd Sell 8,109 (102) UBS Securities Australia	May-14	UBS Securities Australia Ltd	Sell	54	(9)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 194 (94) UBS Securities Australia Ltd Sell 792 (88) UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,909 (712) UBS Securities Australia Ltd Sell 1,909 (212) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 8,109 (45) UBS Securities Australia Ltd Sell 8,109 (102) UBS Securities Australia Ltd Sell 8,109 (145) UBS Securities Australia Ltd Sell 8,69 (44) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	2,702	(300)	Ordinary
UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 792 (88) UBS Securities Australia Ltd Sell 1,70 (13) UBS Securities Australia Ltd Sell 1,006 (1,12) UBS Securities Australia Ltd Sell 1,909 (2,12) UBS Securities Australia Ltd Sell 7,820 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 504 (56) UBS Securities Australia Ltd Sell 504 (45) UBS Securities Australia Ltd Sell 8,109 (102) UBS Securities Australia Ltd Sell 8,109 (145) UBS Securities Australia Ltd Sell 8,109 (145) UBS Securities Australia Ltd Sell 8,109 (145) UBS Securities Australia Ltd Sell 8,61 (44) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	1,800	(200)	Ordinary
UBS Securities Australia Ltd Sell 792 (88) UBS Securities Australia Ltd Sell 117 (13) UBS Securities Australia Ltd Sell 1,008 (112) UBS Securities Australia Ltd Sell 1,909 (2 12) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (2 12) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (90) UBS Securities Australia Ltd Sell 1,306 (102) UBS Securities Australia Ltd Sell 1,306 (145) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	846	(94)	Ordinary
UBS Securities Australia Ltd Sell 1.77 (1.3) UBS Securities Australia Ltd Sell 2.702 (300) UBS Securities Australia Ltd Sell 1,509 (2.12) UBS Securities Australia Ltd Sell 7.02 (300) UBS Securities Australia Ltd Sell 7.82 (87) UBS Securities Australia Ltd Sell 1,800 (2.00) UBS Securities Australia Ltd Sell 1,800 (2.00) UBS Securities Australia Ltd Sell 5.04 (56) UBS Securities Australia Ltd Sell 8,109 (102) UBS Securities Australia Ltd Sell 9,169 (145) UBS Securities Australia Ltd Sell 9,109 (145) UBS Securities Australia Ltd Sell 9,169 (145) UBS Securities Australia Ltd Sell 8,66 (94) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,67 (64) UBS Securities Australia Ltd	Mav-14	UBS Securities Australia Ltd	کو	797	(88)	Ordinan
UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,008 (112) UBS Securities Australia Ltd Sell 1,009 (112) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 8,109 (56) UBS Securities Australia Ltd Sell 8,109 (91) UBS Securities Australia Ltd Sell 8,109 (91) UBS Securities Australia Ltd Sell 1,306 (145) UBS Securities Australia Ltd Sell 8,109 (91) UBS Securities Australia Ltd Sell 8,67 (45) UBS Securities Australia Ltd Sell 8,67 (45) UBS Securities Australia Ltd Sell 8,67 (45) UBS Securities Australia Ltd	May-14	I IBS Securities Australia 14d	100	117	(20)	1000
UBS Securities Australia Ltd Sell 1,702 (300) UBS Securities Australia Ltd Sell 1,009 (712) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 7,802 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 5,64 (46) UBS Securities Australia Ltd Sell 8,109 (40) UBS Securities Australia Ltd Sell 8,109 (45) UBS Securities Australia Ltd Sell 8,109 (45) UBS Securities Australia Ltd Sell 8,409 (45) UBS Securities Australia Ltd Sell 8,409 (45) UBS Securities Australia Ltd Sell 8,600 (45) UBS Securities Australia Ltd Sell 8,600 (45) UBS Securities Australia Ltd		Late of the state	100	711	(600)	Garaly
UBS Securities Australia Ltd Sell 1,008 (112) UBS Securities Australia Ltd Sell 1,008 (112) UBS Securities Australia Ltd Sell 2,702 (300) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,65 (45) UBS Securities Australia Ltd Sell 8,109 (901) UBS Securities Australia Ltd Sell 8,109 (901) UBS Securities Australia Ltd Sell 8,109 (142) UBS Securities Australia Ltd Sell 8,61 (143) UBS Securities Australia Ltd Sell 8,61 (94) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd	lvidy :	Obs securities Australia Ltd	120	7,702	(300)	Ordinary
UBS Securities Australia Ltd Sell 1,509 (212) UBS Securities Australia Ltd Sell 7.02 (300) UBS Securities Australia Ltd Sell 783 (87) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,80 (50) UBS Securities Australia Ltd Sell 8,109 (91) UBS Securities Australia Ltd Sell 8,109 (90) UBS Securities Australia Ltd Sell 1,306 (145) UBS Securities Australia Ltd Sell 1,306 (145) UBS Securities Australia Ltd Sell 8,40 (94) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,69 (94) UBS Securities Australia Ltd	Ividy-14	Obsideration Australia Ltd	Nell Nell	800,1	(112)	Ordinary
UBS Securities Australia Ltd Sell 7.02 (300) UBS Securities Australia Ltd Sell 7.83 (87) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 5,04 (56) UBS Securities Australia Ltd Sell 8,109 (90) UBS Securities Australia Ltd Sell 9,18 (102) UBS Securities Australia Ltd Sell 8,46 (94) UBS Securities Australia Ltd Sell 8,46 (94) UBS Securities Australia Ltd Sell 8,66 (94) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,67 (63) UBS Securities Australia Ltd Sell 8,67 (64) UBS Securities Australia Ltd Sel	Iviay-14	Ubs securities Australia Ltd	Sell	606,1	(212)	Ordinary
UBS Securities Australia Ltd Sell 783 (87) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 135 (15) UBS Securities Australia Ltd Sell 764 (56) UBS Securities Australia Ltd Sell 764 (56) UBS Securities Australia Ltd Sell 8,109 (901) UBS Securities Australia Ltd Sell 9,18 (102) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell <	May-14	UBS Securities Australia Ltd	Sell	2,702	(300)	Ordinary
UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,800 (200) UBS Securities Australia Ltd Sell 1,50 (50) UBS Securities Australia Ltd Sell 5,60 (56) UBS Securities Australia Ltd Sell 8,109 (90) UBS Securities Australia Ltd Sell 1,306 (145) UBS Securities Australia Ltd Sell 8,406 (145) UBS Securities Australia Ltd Sell 8,406 (145) UBS Securities Australia Ltd Sell 8,610 (145) UBS Securities Australia Ltd Sell 8,610 (145) UBS Securities Australia Ltd Sell 8,61 (145) UBS Securities Australia Ltd Sell 8,61 (94) UBS Securities Australia Ltd Sell 8,61 (94) UBS Securities Australia Ltd Sell 8,62 (94) UBS Securities Australia Ltd Sell (85) (85) UBS Securities Australia Ltd	May-14	UBS Securities Australia Ltd	Sell	783	(87)	Ordinary
UBS Securities Australia Ltd Self 1,800 (200) UBS Securities Australia Ltd Self 135 (15) UBS Securities Australia Ltd Self 504 (56) UBS Securities Australia Ltd Self 8,109 (90) UBS Securities Australia Ltd Self 918 (102) UBS Securities Australia Ltd Self 918 (145) UBS Securities Australia Ltd Self 846 (44) UBS Securities Australia Ltd Self 846 (94)	May-14	UBS Securities Australia Ltd	Seli	1,800	(200)	Ordinary
UBS Securities Australia Ltd Sell 135 (15) UBS Securities Australia Ltd Sell 504 (56) UBS Securities Australia Ltd Sell 765 (89) UBS Securities Australia Ltd Sell 8,109 (901) UBS Securities Australia Ltd Sell 9,18 (102) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell (82) (83)	May-14	UBS Securities Australia Ltd	Sell	1,800	(200)	Ordinary
UBS Securities Australia Ltd Sell 504 (56) UBS Securities Australia Ltd Sell 765 (85) UBS Securities Australia Ltd Sell 8109 (90) UBS Securities Australia Ltd Sell 1,306 (145) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 867 (63) UBS Securities Australia Ltd Sell 846 (94)	May-14	UBS Securities Australia Ltd	Sell	135	(15)	Ordinary
UBS Securities Australia Ltd Sell 765 (85) UBS Securities Australia Ltd Sell 8,109 (901) UBS Securities Australia Ltd Sell 1,306 (102) UBS Securities Australia Ltd Sell 8,46 (94) UBS Securities Australia Ltd Sell 846 (94)	May-14	UBS Securities Australia Ltd	Sell	504	(95)	Ordinary
UBS Securities Australia Ltd Sell 8.109 (91) UBS Securities Australia Ltd Sell 9.18 (102) UBS Securities Australia Ltd Sell 9.18 (102) UBS Securities Australia Ltd Sell 8.46 (9.4) UBS Securities Australia Ltd Sell 8.67 (6.3) UBS Securities Australia Ltd Sell 8.46 (9.4)	May 14	I IBS Societies Australia 11d	100	365	(00)	Ording)
UBS Securities Australia Ltd Sell 0,109 190 UBS Securities Australia Ltd Sell 0,102 102 UBS Securities Australia Ltd Sell 1,306 (145) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 866 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94)	May 14	1 DC Coomition Australia ted	120	507	(600)	Guinary
UBS Securities Australia Ltd Sell 1,306 (102) UBS Securities Australia Ltd Sell 1,306 (143) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 567 (63) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (34) UBS Securities Australia Ltd Sell 846 (34)	Widy-14	Saculinas	Seli	8) l/8	(201)	Ordinary
UBS Securities Australia Ltd Sell 1,306 (145) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 867 (63) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (32)	May-14	UBS Securities Australia Ltd	Sell	918	(102)	Ordinary
UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 567 (63) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 198 (22)	May-14	UBS Securities Australia Ltd	Sell	1,306	(145)	Ordinary
UBS Securities Australia Ltd Sell 567 (63) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94)	May-14	UBS Securities Australia Ltd	Sell	846	(94)	Ordinary
UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 198 (22)	May-14	UBS Securities Australia Ltd	Seil	295	(63)	Ordinary
UBS Securities Australia Ltd Sell 846 (94) UBS Securities Australia Ltd Sell 198 (22)	Mav-14	UBS Securities Australia Ltd	llay	846	(00)	Ordinary
UBS Securities Australia Ltd Sell (22)	May 14	198 Securities Australia 14d	1100	949	(10)	(Bullion)
UBS Securities Australia Ltd Sell (22)	May-1-	ODS Securices Mostiglia Ltd	liac	040	1000	C PERIONS
	A Anni 7 a		-		, ,	

Self 2,475 19,137 10,1	(272) Ordinary				(200) Ordinary		(390) Ordinary			(53) Ordinary	l	(300) Ordinary					(640) Ordinary	(133) Ordinary		301) Ordinary		(153) Ordinary		210) Ordinary				(11) Ordinary	L	(13) Ordinary			(347) Ordinary		(15) Ordinary		(474) Ordinary		(735) Ordinary		241) Ordinary		(502) Ordinary	(393) Ordinary				(200) Ordinary	Ordinary Ordinary			(3,181) Ordinary				
		(2)		(2)									-																												-															
				7					ľ	Ì				3									-																							-	1		I.P	101		28				E 6
securities Australia Ltd Securities Australia	Sell	Seli	Seil	Sell	Sell	Sell	Sell	yell	Zell	Sell	Sell	Seli	Sell	les.	Sell	Sell	Sell Sell	Į 3	Sell	Sell	Sell	Sell	حوا الإ	Sell	Self	Sell	Seli	lac Zella	Sell	Sell	Sell	Sell	10 N	Sell	Sell	Sell	Sell	Ŗ 3	Sell	Sell	Sel	Sell	Sell	Sell	Sell	Sell	Self	Seli	llac Jac	, J	Sell	Sell	Sell	-	ž,	Sell
securities Australia IL Securities Australia II Securities Australia II Securities Australia II Securities Australia II Securities Australia II Securi	P	P	P	9	· O	9	0				0	P	P	d	q	ď	, i		F	-			, ,	P	F					-	75.				T	70				70	70					71							7	_		
STANDARD BEFORE STANDARD STAND	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia L	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia 11	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lt	Securities Australia Lto	Securities Australia Lto	Securities Australia Lto	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Lit	Securities Australia Lt	Securities Australia Lto	Securities Australia Lto	UBS Securities Australia Ltd	Consisting Assertalia I to	Securities Australia	UBS Securities Australia Ltd

					2
May-14	UBS Securities Australia Ltd	Sell	855	(36)	Ordinary
08-May-14	UBS Securities Australia Ltd	Self	1,647	(183)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	66	(11)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	000'6	(1,000)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	1,647	(183)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	2,304	(526)	Ordinary
38-May-14	UBS Securities Australia Ltd	Self	189'651	(17,743)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sett	1,305	(145)	Ordinary
38-May-14	UBS Securities Australia Ltd	Sell	800'61	(2,112)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	1,827	(203)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	24,183	(2,687)	Ordinary
08-May-14	UBS Securities Australia Ltd	-BS	286	(104)	Ordinary
08-May-14	UBS Securities Australia Ltd	-Se	9.486	(1 054)	Ordinary
8-May-14	UBS Securities Australia Ltd	Sell	64 296	(7.144)	Ordinary
May-14	I IBS Securities Australia I to	190	0 101	(000)	Ordinary
O8 May 14	I IRC Socration Ametralia 14d	2011	6,161	(202)	Ordinary
Mary	A 100 Committees Australia Ltd	100	100'5	(5/0,1)	Oldinary
t-inidy-14	OBS Securities Abstralia Ltd	liac l	18,11/	(2,013)	Ordinary
US-IMay-14	UBS Securities Australia Ltd	Jell Sell	819	(16)	Ordinary
Jo-May-14	UDS SECURIES AUSTRAIN LTG	100	15,894	(99/'1)	Ordinary
8-May-14	UBS Securities Australia Ltd	lie.	113,256	(12,584)	Ordinary
08-May-14	UBS Securities Australia Ltd	Sell	270,000	(30,000)	Ordinary
May-14	UBS AG London Branch	Sell	417,076	(46,035)	Ordinary
9-May-14	UBS AG London Branch	Buy	418,444	46,035	Ordinary
39-May-14	UBS Securities Australia Ltd	Sell	17,743	(1.953)	Ordinary
9-May-14	UBS Securities Australia Ltd	Sell	73.133	(8,030)	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Silv	164 073	18.030	Ordinary
19-May-12	I BS Securities Australia 11d	Biny	340 004	40,410	Common
1	COO DECALLIES AUSTRAIN LIU	ina .	449,030	49,419	Orginary
3-May-14	UBS Securities Australia Ltd	Seil	581,384	(63,960)	Ordinary
39-May-14	UBS Securities Australia Ltd	Buy	418,444	46,035	Ordinary
09-May-14	UBS Securities Australia Ltd	Buy	152,610	16,800	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Sell	351 346	(38,800)	Ordinary
19-May-14	LIRS Sequities Australia Ltd	Zell	100	(11, 503)	ordina O
May 14	LIBS Countries Australia 14d	0.00	42,001	(666,11)	Ordinary
	COO CONTRACTOR COO	én c	- Diff	200,4	Cigniany
9-May-14	UBS Securities Australia Ltd	New Year	145,318	(15,985)	Ordinary
9-May-14	UBS Securities Australia Ltd	Sell Sell	763,772	(84,015)	Ordinary
9-May-14	UBS Securities Australia Ltd	Enter Into Derivative (A/N	103,484	Options
May-14	UBS Global Asset Management	Buy	2.317	254	Ordinary
09-May-14	ILIBS Wealth Management Australia Ltd	Riv	10,000	1 1 40	Ordinan
00 May 14	LIDE Committee Australia 144	Coll	12,121	(00)	O'dillially
1 :	COS Securities Australia Liu	laci	0,45/	(90/)	Oraniary
JS-May-14	UBS Securities Australia Ltd	Sell	6,229	(683)	Ordinary
39-May-14	UBS Securities Australia Ltd	Sell	285,483	(31,303)	Ordinary
19-May-14	UBS Securities Australia Ltd	ll as	9.868	(1.082)	Ordinary
09-May-14	I IBS Securities Australia 14d	B. I.	120	(700-1)	Ordinary
1	ODD Securities Australia Ltd	, and	120	4	Orginary
09-May-14	UBS Securities Australia Ltd	Sell	95,040	(10,421)	Ordinary
39-May-14	UBS Securities Australia Ltd	Sell	2,316	(254)	Ordinary
79-May-14	ILIBS Securities Australia Ltd	le?	118	(13)	Ordinan
	T-1 - 1 - 4 1 - 1 - 2 - 2 - 2 - 2 - 2 - 2 - 2 -		2	(C.)	Ciciliary
J9-May-14	UBS Securities Australia Ltd	Nell Nell	219	(24)	Ordinary
/ay-14	UBS Securities Australia Ltd	Sell	5,985	(657)	Ordinary
39-Mav-14	UBS Securities Australia I td	Sel	465	(F.1)	Ordinary
00 kd 20, 1.4	PiDC Committee Aumentical ed	1-3	100.	10000	
vidy- i	ODS SECULIES AUSTRAIN L'U	oeii	/57,1	(138)	Ordinary
09-May-14	UBS Securities Australia Ltd	Sell	146	(16)	Ordinary
Vay-14	UBS Securities Australia Ltd	- Sell	191	(121)	Ordinary
39-May-14	IUBS Securities Australia Ltd	Sell	46	(5)	Ordinary
11	LIBO Commission Assessed and		000	(011)	
39-Ividy-14	ODS SECULIES AUSTRAINA LIC	Jell Sell	4,282	(4/0)	Orginary
May-14	UBS Securities Australia Ltd	Sell	6,495	(713)	Ordinary
.09-May-14	UBS Securities Australia Ltd	Sell	6,195	(089)	Ordinary
14-ve/	I IRS Securities Australia 11d	J. J.	10 732	(1 179)	Ordina
39-IVI dy- 14	Upo Securities Australia Ltd	Ser.	10,73	(0/1/1)	Cramary
/ay-⊥4	UBS Securities Australia Ltd	Seli	5,284	(280)	Ordinary
99-May-14	UBS Securities Australia Ltd	Sell	2,095	(230)	Ordinary
09-May-14	UBS Securities Australia Ltd	Sell	337	(32)	Ordinary
09-May-14	LIBS Securities Australia Ltd	Sell	3 781	(415)	Ordinary
Any 1 A	100 Cocharton Australia Lad	1100	202	(50)	O'CHING)
+1-4pin-60	OD Seculities Australia Lto	3411	567	(/0)	Ordinary
May-14	UBS Securities Australia Ltd	Sell	1,239	(136)	Ordinary
39-May-14	UBS Securities Australia Ltd	Sell	911	(20)	Ordinary
May-14	UBS Securities Australia Ltd	Sell	6.523	(716)	Ordinary
				(01.)	Citation
		-			

	The second secon	_	0000		
09-May-14	UBS Securities Australia Ltd	Sell	219	(1,4(7)	Ordinary
09-May-14	Securities	Sell	13.765	(1.511)	Ordinary
09-May-14	UBS Securities Australia Ltd	Seil	6,450	(208)	Ordinary
09-May-14	UBS Securities Australia Ltd	las	3,772	(414)	Ordinary
09-May-14	UBS Securities Australia Ltd	les	4 214	(460)	Ordinary
09-May-14	UBS Securities Australia Ltd	Sel	19 133	(100 ()	Ordinary
09-May-14	UBS Securities Australia Ltd	lles	14,228	(1.555)	Ordinary
09-May-14	UBS Securities Australia Ltd	38	17.641	(1.928)	Ordinary
09-May-14	UBS Securities Australia Ltd	3,2	924	(101)	Ordinary
ı	UBS Securities Australia Ltd	Sell	11,474	(1.254)	Ordinary
ı	UBS Securities Australia Ltd	les	2.157	(737)	Ordinary
ı	UBS Securities Australia Ltd	- Jan	828	(16)	Ordinary
1	UBS Securities Australia Ltd	les.	8.268	(116)	Ordinary
12-May-14	UBS Securities Australia Ltd	Buy	149.670	16.525	Ordinary
12-May-14	UBS Securities Australia Ltd	3	45.850	(000.5)	Ordinary
12-May-14	UBS Securities Australia Ltd	lle.	195,040	(2) 564)	Cipio
12-Mav-14	UBS Securities Australia I to	Ruy	171 570	30.200	Ordinary
12-Mav-14	UBS Securities Australia I to	(2)	477,273	607'00	Ordinary
12-Mav-14	UBS Securities Australia I to	Sin	00001	(110)	Ordings
12-Mav-14	UBS Sequities Australia 1td	(Sell	230,651	011,1	Quinary
12-Mav-14	UBS Global Asset Management	Riv	137,880	14 032	Ordinary
12-May-14	UBS Global Asset Management	Buy	46 753	5 176	Ordinary
12-May-14	UBS Wealth Management Australia Ltd	Sell	1,539	(071)	Ordinary
12-May-14	UBS Securities Australia Ltd	Buy	16,535,664	1.828,135	Ordinary
12-May-14	UBS Securities Australia Ltd	Buy	51,859	5,743	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	25,799	(2,857)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	42,757	(4,735)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	18	(6)	Ordinary
12-May-14	UBS Securities Australia Ltd	Seli	8,109	(868)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	108	(12)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	25,311	(2,803)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	6,283	(1,028)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	225,750	(25,000)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	16,254	(1,800)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	3,784	(419)	Ordinary
12-May 14	100 Counties Australia Ltd	100	015,1	(140)	Ordinary
2-May-14	18C Societies Australia 14d		000,070	(965,8)	Ordinary
2-Mav-14	UBS Securities Australia Ltd	19	24.250	(400,00)	Ordinary
2-Mav-14	LIBS Securities Australia Ltd		1 806 000	(000,000)	Ordinary
2-May-14	UBS Securities Australia Ltd	195	361 724	(40.058)	Ordinary
2-May-14	UBS Securities Australia Ltd	9	25,853	(000,04)	Ordinary
2-May-14	UBS Securities Australia I to	J	57,602	(2,003)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	659	(73)	Ordinary
Z-May-14	UBS Securities Australia Ltd	ll as	050.6	(1 000)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	9.283	(1 028)	Ordinary
2-May-14	UBS Securities Australia Ltd	las	451.500	(50,000)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	51,615	(5 716)	Ordinary
2-May-14	UBS Securities Australia Ltd	lleS	26,386	(2,922)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seil	13,906	(1.540)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	19,026	(2.107)	Ordinary
2-May-14	UBS Securities Australia Ltd	Buy	15,271	1,693	Ordinary
2-May-14	UBS Securities Australia Etd	Buy	5,827	949	Ordinary
2-May-14	UBS Securities Australia Ltd	Buy	5,926	657	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	1,363,702	(151,019)	Ordinary
2-May-14	UBS Securities Australia Ltd	Buy	1,633	181	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	22,642	(2,506)	Ordinary
2-May-14	UBS Securities Australia Ltd	.Sell	2,205	(244)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	669'06	(10,032)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	804,991	(89,097)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,355,250	(150,000)	Ordinary
Z-May-14	UBS Securities Australia Ltd	Sell	127	(14)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	470	(52)	Ordinary
2-May-14	UBS Securities Australia Ltd	las.	642	(71)	Ordinary
	UBS Securities Australia Ltd	28	63	(7)	Ordinary
				(447)	

V-May-14	Carlotte Control Control				Secions
-May-14	LIRS Securities Australia 14d	lion of	36	(00)	
12-May-14	Т	3	3 752	(415)	Ordinary Specifically
12-May-14	Ť		30,10	(12)	ie ie
2 May 14	LIBC Cocurities Australia Ltd	100	801	(71)	Orginary
May 14	I IDC Committee Australia Ltd	= 5	500	(60)	Orginary
t (0)4 5		== C	76/1/	(202)	Ordinary
-Ividy-14	- 1	Sell	20,306	(2,245)	Ordinary
2-Ividy-14	UBS Securities Australia Ltd	Sell	30,4/3	(3,369)	Ordinary
2-IVI3y-14	UBS Securities Australia Ltd	Sell	7,815	(864)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	5,201	(575)	Ordinan
2-May-14	~	Sell	452	(20)	Ordinary
2-May-14	UBS	Sell	3,392	(375)	Ordinan
-May-14		Sell	6.485	(717)	Ordinary
2-May-14	Ť	Sell	2 940	(325)	Ordinary
7. May 14	LIBC Commission Australia Ltd	100	2000	(020)	500
1	Library Australia Ltd	ii a	2,201	(052)	Ordinary
-ividy-	OBS Securities Australia Ltd	Seli	7,761	(250)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,330	(147)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	7,815	(864)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,502	(166)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	23,584	(2.606)	Ordinary
2-May-14	UBS Securities Australia Ltd	Jay.	698	(96)	Ordinar
2-May-14	LIBS Securities Australia Ltd	3	30,679	(20, 7)	Grain
May 14	* IDC Counting Australia Ltd	11-00	670,02	(07/7)	Collina
2 May 14	Con Securities Adstralia Lid	II o	90	(a)	Ordinary
-May-14	UBS Securities Australia Ltd	Seil	12,670	(1,400)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seil	050'6	(1,000)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,339	(148)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	13,575	(1,500)	Ordinary
-May-14	UBS Securities Australia Ltd	95	3.593	(397)	Ordinar
-Mav-14	l l	9	35	(4)	Ording
2-May-14	Later Continue Contin	- 3	0,0	(t)	500
		180	Or's	(525)	Coural
-Ividy-14	- 1	Sell	82	(6)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	7,801	(862)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	21,087	(2,330)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	36.707	(4.056)	Ordinar
2-May-14	UBS Securities Australia Ltd	Sell	4.815	(532)	Ordinary
-Mav-14	ILIBS Securities Australia 11d	lla)	21811	(0.10)	Colora
2 May 14	Libe Committee Australia 144	==0	0.00	(1,00)	5 0
-APIA-	ODS SECULINES AUSTRAINA LIU	Sell	969'6	(579)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	11,313	(1,250)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seil	50,047	(5,530)	Ordinan
2-May-14	UBS Securities Australia Ltd	Sell	4,625	(511)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	253	(28)	Ordinan
2 Mary 14	LOC Commission Assertation 140		0200	(07)	i i
May-14	Obs securities Australia Ltd	Sell	050/6	(000,1)	Ordinary
Z-May-14	UBS Securities Australia Ltd	Sell	833	(95)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	7,801	(862)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	10,770	(061-1)	Ordinan
-May-14	URS Securities Australia I to	[ay	22	(3)	Color
May 14	1 IDC Countries Australia Ltd		/2	(C)	- A
12-IVIdy-14	Obs Securities Australia Ltd		019'/	(840)	Ordinary
-May-14	UBS Securities Australia Ltd	lles.	3,651	(403)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,368	(151)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	834	(65)	Ordinary
2-Mav-14	UBS Securities Australia Ltd	Şe	1 024	(113)	Ordinar
2-May-14	185 Sourrities Australia 11d	Coll	7,007	(000)	in in
14	Late attended to see a second		200,1	(1004)	
+1-kpin-7	ODS SECURITIES AUSTRALIA LLG	Sell	72,393	(086'/)	Orginary
-May-14	UBS Securities Australia Ltd	Sell	29,445	(3,250)	Ordinary
12-May-14	UBS Securities Australia Ltd	=se	924	(102)	Ordinary
2-May-14	II IRS Securities Australia I to	las	A 206	(454)	Ording
	100 Comments A 100 COM	====	202,4	(101)	E 15 C
4-1-ADIA1-7	Obsideral ties Australia Ltd	Jac Jac	676	(9c)	Orginary
2-May-14	UBS Securities Australia Ltd	Sell	5,726	(632)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	63	6	Ordinary
2-May-14	UBS Securities Australia Ltd	Self	3.869	(427)	Ordinary
2-May-14	UBS Securities Australia Ftd	lles	500	(44)	Ordinan
2-May-14	IBS Securities Australia Ltd	05	7 302	(808)	Ordinan
2-May-14	I IDC Coounting Australia 14d	= 0	700	(000)	in in
2 145	100 Commission Assessed Land	1123		2 5	Cidinal
rvidy	UBS Securities Australia Ltd	Sel	. 50		CO CONTRACTOR
				(a)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	4,285	(b) (473)	Ordinary

2-May-14 UBS Securities Australia Ltd 2-May-14 UBS Securities Australi		4,465 5,334 5,334 2,073 1,618 11,300 11,300 13,560 13,	(1,500) (1,500	Ordinary Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		2,079 9,736 9,736 1,518 1,130 1,112 226 705 1,112 221,120 13,560 14,500 15,500 16	(1792) (1793) (1	Ordinary Ordinary
3		9,736 9,736 18,080 18,080 17,319 17,319 17,310 17,310 17,320 1,320 18,236 1	(1,07) (1	Ordinary Ordinary
		9,736 9,736 1,618 11,300 11,300 13,560 1	(1,27) (1,250) (1,500)	Ordinary Ordinary
		1,130 1,130 1,130 1,130 1,130 1,112 1,112 1,112 1,120 1,136 1,130 1,	(173) (1,500)	Ordinary Ordinary
_		1618 162080 77,319 77,319 705 705 705 705 705 13,560 13,792 77,120	(17.9) (1.250)	Ordinary Ordinary
		18,080 17,319 17,319 17,32 226 705 1,112 22,120 13,560 14,700 14,	(2,503) (1,503	Ordinary Ordinary
		77,319 11,300 7,792 226 7,792 13,560	(8.53) (1.250) (1.250) (1.750)	Ordinary Ordinary
		11,300 7792 226 7792 266 11,12 27,120 13,560 18,560 18,560 18,560 18,560 19,237 19,237 11,300	(1,250) (1,500	Ordinary Ordinary
		7,792 705 705 705 705 13,500 13,500 18,560 13,560 14,560 1	(1500) (1500) (1700) (1	Ordinary Ordinary
		7.52 7.05 7.05 1.112 2.7,120 1.1846 1.18	(32) (32) (32) (32) (32) (32) (32) (32)	Ordinary Ordinary
		705 1,112 27,120 13,560 13,560 16,336 16,336 17,92 17,92 17,92 17,120 17,120 17,120 17,120 11,300 11	(37) (1500) (1,500) (1	Ordinary Ordinary
3		1,112 27,120 13,560 13,560 18,236 18,244 13,560 13,	(3.78) (3.000) (1,500)	Ordinary Ordinary
		27,112 27,120 13,560 16,236 16,236 1,844 13,560 14,730 11,300 11,	(3200) (1,500)	Ordinary Ordinary
- 1 1 1 1 1 1 1 1 1 1		2,120 13,560 16,236 16,236 17,92 17,92 17,92 17,92 17,00 11,300 1	(1,500) (1,500	Ordinary Ordinary
_		13,560 15,260 16,236 18,236 13,560 13,560 13,560 13,560 13,560 13,560 13,560 13,560 13,560 13,560 13,560 13,560 13,560 13,560 13,560 11,300 11	(1,500) (1,786) (1,786) (1,786) (1,500	Ordinary Ordinary
		13.560 16.236 18.44 18.44 13.560 13.560 13.560 19.237 19.237 19.237 11.300 27.120 27.120 26.50 26.50 26.50 26.50 27.720 27.720 27.720 27.720 27.720	(1,500) (1,500	Ordinary Ordinary
		16.236 1.1844 1.18560 1.18560 1.18560 1.18560 1.1860 27,120 27,120 27,120 27,120 28,520 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,650 2,792 7,792 7,792	(1,296) (1,200	Ordinary Ordinary
		1,844 13,560 7,792 13,560 13,560 13,560 22,120 22,120 26,120 26,120 26,120 26,120 26,120 26,120 26,120 26,120 27,120 28,130 11,300 11,300 11,300 27,792 7,792	(1,500) (1,500	Ordinary Ordinary
		13,560 13,560 13,560 13,560 13,560 13,000 11,300 22,650 5,650 5,650 11,300 11,3	(1,500) (1,500	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
		7,792 13,560 13,560 19,237 11,300 27,120 25,50 26,62 5,650 11,300 11,300 8,6874 7,792 7,792	(862) (1,200) (3,000) (3,000) (3,000) (3,000) (3,000) (3,000) (3,000) (3,000) (4,000) (1,000)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
3		13,560 13,560 13,560 22,1,20 11,300 22,1,20 26,50 5,650 5,650 5,650 11,300 11,3	(328) (1005(1) (1005(Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
		13.560 13.560 19.237 11.300 27.120 27.120 26.550 5.650 5.650 11.300 11.300 86.74 77.72 77.92	(1,500) (1,500) (2,128) (1,250) (8,300) (8,250) (6,250) (1,260) (1,260) (1,260) (1,260) (1,260) (1,260)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
_		13,560 19,237 19,237 11,300 22,120 5,650 5,650 11,300 11,300 86,874 7,792 7,792	(1,500) (2,120) (2,120) (1,250) (8,425) (2,50) (1,000) (1,250) (1,000) (1,250) (2,50) (1,250) (3,000) (1,250) (1,250) (1,250)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
		27,120 19,237 11,300 27,120 76,120 76,120 5,650 9,040 11,300 11,300 88,74 7,792 7,792	(3,000) (2,128) (1,250) (3,000) (625) (1,000) (1,250) (1,250) (1,250) (1,250) (1,250) (1,250)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
		19,237 11,300 27,120 27,120 5,650 5,650 5,650 10,300 11,300 86,874 7,792 7,792	(2,128) (1,250) (3,000) (8,425) (625) (1,000) (1,250) (1,250) (9,610) (9,610)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
		11,300 27,120 27,120 76,120 5,650 2,5650 9,040 11,300 11,300 86,874 7,792 7,792	(1,250) (3,000) (8,425) (625) (1,000) (1,250) (1,250) (9,610)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
		27,120 76,162 5,650 5,650 2,850 1,040 11,300 86,874 7,792	(3,000) (8,425) (625) (1,000) (1,250) (1,250) (9,610)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
		76,162 5,650 226 9,040 11,300 86,814 7,792	(8,425) (8,425) (25) (1,000) (1,250) (1,250) (1,250) (9,610)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
		5,555 5,556 226 9,040 11,300 8,874 7,792	(625) (625) (1,000) (1,250) (1,250) (1,250)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	226 226 9,040 11,300 86,874 7,792	(255) (255) (1,000) (1,250) (9,610)	Ordinary Ordinary Ordinary Ordinary Ordinary
		226 9,040 11,300 86,874 7,792	(25) (1,000) (1,250) (9,610)	Ordinary Ordinary Ordinary Ordinary
	See 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	9,040 11,300 86,874 7,792 750	(1,000) (1,250) (9,610)	Ordinary Ordinary Ordinary
		11,300 86,874 7,792 750	(1,250)	Ordinary Ordinary Ordinary
	Sel	86,874 7,792 750	(9,610)	Ordinary Ordinary
	Sell Sell	7,792	(862)	Ordinary
	Sell Sell	750		
	lles lles	S,	(60/	Ordinan
	100	202	(50)	Cidinaly
	9	900	(ac)	Clustery
		44,358	(4,908)	Ordinary
	Seli	206	(99)	Ordinary
	Sell	461	(51)	Ordinary
	Sell	22,916	(2,535)	Ordinary
	lleS	172	(10)	Ordinary
	100	030 C	(604)	Circle of the ci
	120	006,6	(430)	Ordinary
	Sell	5,424	(009)	Ordinary
	Sell	11,300	(1,250)	Ordinary
	Sell	1,772	(196)	Ordinary
1	- Se	913	(101)	Ordinary
	100	0000	(00)	Ordinan,
	i i	5,025	(770)	Ordinary
11111	Sell	34,804	(3,850)	Ordinary
	Sell	1,772	(196)	Ordinary
1 1 1 1 1 1	Sell	695'5	(616)	Ordinary
	Sell	36,160	(4.000)	Ordinary
1111	les	3,679	(407)	Ordinary
	Coll	027 12	(2000)	Ordina.
8 8 8 8	ier i	0/4,17	(006,7)	Collialy
	Sell	75,564	(2,496)	Ordinary
SB) SB)	Sell	123,966	(13,713)	Ordinary
OBS	Sell	362	(40)	Ordinary
	Sell	54.584	(6.038)	Ordinary
ĭ	Coli	15 000	(0.00(0)	Ordings
Ť	11-0	686,61	(00/1)	Olumary
7	Seil	797	(67)	Ordinary
2-May-14 UBS Securities Australia Ltd	Self	5,297	(286)	Ordinary
	Sell	18	(2)	Ordinary
Т	Sell	90.400	(10,000)	Ordinary
T	197	17,700	(1,058)	Ordinan
Ť	190	20/1/	(006,1)	, oldinary
1	Sell	5,496	(809)	Ordinary
2-May-14 UBS Securities Australia Ltd	lles.	72	<u>(8)</u>	Ordinary
Т	les	163	(18)	Ordinary
1	2611	601	(0)	Clairiary

2-may-14	UBS Securities Australia Ltd	Sell	896'6	5	Ordinary
12 May-14	UBS Securities Australia Ltd	- R	866,59	٥	Ordinary
12-May-14	Libs countries Australia Ltd	3el	015,7	(878)	Ordinary
12 1/43 14	OBS Seculities Advisagle Cid	lak lak	3,528		Ordinary
12-IVI29-14	UBS Securities Australia Ltd	lax.	6,/84		Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	880		Ordinary
12-May-14	UBS Securities Australia Ltd	-Se	1,669	(184)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	3,655	(403)	Ordinary
12-May-14	UBS Securities Australia Ltd	Zell	8.471	(786)	Ordinary
12-May-14	It IBS Sequitities Australia Ltd	Cell	060	(100)	in in in
12-May-14	I IDC Cocumition Australia 1+4	200	200	(100)	O G
	COS CONTINUES COST COST COST COST COST COST COST COS	361	0000	(ccc)	Ordinary
7-May-14	UBS Securities Australia Ltd	les	8,335	(919)	Ordinary
12-May-14	UBS Securities Australia Ltd	Seli	490	(54)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	762	(84)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	499	(22)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	8.154		Ordinary
12-May-14	UBS Securities Australia Ltd	وا	7.5	(6)	Croins
12 May 14	I IDC Cocurision Australia Ltd	100	/2	(6)	S G
12 1424 14	and the state of t	30:1	*^	(a)	Ordinary
12-IVIdy-14	Obs securities Australia Ltd	Seil	1,73	(081)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	4,535	(200)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	13,605		Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	0/0/6	(1)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	8/1/8		Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	8.163	(006)	Ordinary
12-Mav-14	UBS Securities Australia Ltd	es.	5 687		Ordinary
12-May-14	ILIBS Securities Australia 14d	J.	11 330		
2-May-14	1 IDC Countries Australia 14d	100	060	(355)	Gurary
-1- (pla) -1-	מביתו ונובי שתפתפוום בית	lac.	0,340		Ordinary
Z-IViay-14	UBS Securities Australia Ltd	ie.	20,924		Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	6,803	(220)	Ordinary
12-May-14	UBS Securities Australia Ltd	38	020'6		Ordinary
12-May-14	UBS Securities Australia Ltd	les.	2,268		Ordinary
2-May-14	I IRS Securities Australia 14d	Coll	735.4		
2 May 14	CIDS Countries Australia 14d	100	1000		Ciginal Ciginal
2 - Ividy 14	Opp Securities Australia Lid	i A	/06	(001)	Ordinary
2-May-14		, Š	33,559	(3,700)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	16,136	(1,779)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,868	(907)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,215	(134)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	444	(49)	Ordinary
2-May-14	I IBS Securities Australia I to	Call	966		Codings
2 140.1		100	OCC.		Collialy
4-14ldy-14	ODS SECULIFIES AUSTRAIIA LIU	nac .	8/5/8	(950'T)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	32,652		Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	18362		Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	6.521		Ordinary
2-May-14	1 IBS Sequitities Australia 14d	Coll	200 0		Ording)
7 140, 14	The securities Australia Liu	200	/00'0		Ordinary
4-May-14	ODS SECULIES AUSITALIA LIG	Sell	064	(54)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	245	(22)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	190	(12)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	3.964	(437)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sol	163	(18)	Croins
2-May-14	I IBS Securities Australia 144	100	001	(6.7)	Ordinary.
2-May 14	I IDC Countries Australia 144	100	9	(10)	9
2-Ividy-14	ODS Securities Australia Ltd	lak.	8	(7)	Orginary
Z-May-14	UBS Securities Australia Ltd	Sell	3,519	(388)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	18	(2)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seli	1.034	(114)	Ordinary
2-May-14	URS Sequirities Australia Ltd	Coll	8678	(705)	Colingo
2. May 1.4	100 Committee Australia 140	No.	566	(1)	1
2-ividy-14	Obs seculities Australia Ltu	lac d	60	(/)	Ordinary
2-way-14	Obs Securities Australia Ltd	Seli	869	(//)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seli	7,202	(794)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	798	(88)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	2,857	(315)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	689	(94)	Ordinary
2-May-14	I IRS Securities Australia I to	Sall	15 446	(5021)	Ordinan
	The state of the s	100	Oht/C	(00/1)	, ai
2-IVIGY-14	Obs securities Australia Lid	Sell	667	(33)	Ordinary
Z-May-14	UBS Securities Australia Ltd	Sell	13,542	(1,493)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	2,304	(254)	Ordinary
2-Mav-14	UBS Securities Australia Ltd	lle's	2 231	(346)	Ordinary
, (nin, 2	כמון מבייתו וונבים שיחים וומום דיירים	Jaci	1,777	(U#2)	Clumary
12-May-14	UBS Securities Australia Ltd	Sell	80	(2)	Ordinary

UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell 113 UBS Securities Australia Ltd Sell 114 UBS Securities Australia Ltd Sell 115 UBS Securities Australia Ltd Sell 22 UBS Securities Australia Ltd Sell 25 UBS Securities Australia Ltd Sell 26 UBS Securities Australia Ltd Sell 26 <t< th=""><th></th><th>BS Securities Australia Ltd</th><th>Sell</th><th>236</th><th>(26)</th><th>Ordinary</th></t<>		BS Securities Australia Ltd	Sell	236	(26)	Ordinary
USS Securities Australia Ltd Sell 1,1296 1,1297 1,1498			0-11	r cr	1,00	
USS Securities Australia Ltd Sell 11,378 1,439		BS Securities Australia Ltd	Sell	734	(81)	Ordinary
USS Securities Australia Ltd Sell 17.2		BS Securities Australia Ltd	Sell	13 037	(4)	
USS Securities Autorialia Ltd Sell 172 172 173 173 174 174 175 174 175 1		RS Societies Australia Ltd	Call	(50/61	(10/4)	
USS Securities Australia Ltd Sell 1,1249		DC Coo without Australia 144	100	177	(34)	Ordinary
USS Securities Australia Ltd Sell 1,276 1,270 1,000	1	PC Contraction Australia 144	100	7/1	(1900)	Colliary
UBS Securities Australia Ltd Sell 1,236 (1,247) (188) (1985 1,247) (188) (1985 1,247) (1985 1,247) (1985 1,247) (1985 1,247) (1985 1,247) (1985 1,247) (1986 1,247)		DO Securities Australia Ltd	100	non's	(000)	Ordinary
USS Securities Australia Ltd Sell 17579 17571 1757 17571 1757 17571 1757 17571 1	Т	DC Committee Australia Ind	100	600 11		l
USS Securities Australia Ltd Sell 2,5513 1,756 1,595 1,985	T	So Securities Australia I to	روا	702		
USS Securities Australia Ltd Sell Sell 17,76 10,50	T	DC Contribute Australia 144	100	151		Ordinary
UBS Securities Australia Ltd Sell 1,776 (196) UBS Securities Australia Ltd Sell 1,776 (196) UBS Securities Australia Ltd Sell 2,513 (2,816) (196)	1	DC Cocurities Australia Ltd	100	507'7		
Cost Securities Australia Ltd Sell 1,23,51 (1,56) UBS Securities Australia Ltd Sell 7,810 (1,96) UBS Securities Australia Ltd Sell 3,650 (1,09) UBS Securities Australia Ltd Sell 3,650 (1,09) UBS Securities Australia Ltd Sell 3,650 (1,09) UBS Securities Australia Ltd Sell 1,016 (1,121) UBS Securities Australia Ltd Sell 1,016 (1,121) UBS Securities Australia Ltd Sell 1,016 (1,121) UBS Securities Australia Ltd Sell 1,016 (1,131) UBS Securities Australia Ltd Sell 8,916 (1,056) UBS Securities Australia Ltd Sell 8,916 (1,056) UBS Securities Australia Ltd Sell 8,916 (1,056) UBS Securities Australia Ltd Sell 8,920 (1,056) UBS Securities Australia Ltd Sell 8,920 (1,056) UBS Securities Australia Ltd Sell 8,920 (1,056)	+	Do Securities Australia Liu	liac liac	70,700	(0)	
UBS Securities Australia Ltd Sell 7.37 (1.96) UBS Securities Australia Ltd Sell 8.95 (2.91) UBS Securities Australia Ltd Sell 8.95 (4.02) UBS Securities Australia Ltd Sell 3.61 (3.98) UBS Securities Australia Ltd Sell 3.61 (3.98) UBS Securities Australia Ltd Sell 3.61 (3.98) UBS Securities Australia Ltd Sell 3.62 (4.22.1) UBS Securities Australia Ltd Sell 3.66 (4.08) UBS Securities Australia Ltd Sell 3.68 (4.98) UBS Securities Australia Ltd Sell 3.68 (4.98) UBS Securities Australia Ltd Sell 3.68 (4.98) UBS Securities Australia Ltd Sell 3.59 (5.90) UBS Securities Australia Ltd Sell 4.535 (5.90) UBS Securities Australia Ltd Sell 4.535 (5.90) UBS Securities Australia Ltd Sell 4.535 (5.90) UBS Securities Australi	+	po securities Australia Ltu	riac I	25,6,62		
UBS Securities Australia Ltd Sell 7.510 (1662) UBS Securities Australia Ltd Sell 3.610 (399) UBS Securities Australia Ltd Sell 1.016 (1.213) UBS Securities Australia Ltd Sell 1.016 (1.234) UBS Securities Australia Ltd Sell 8.560 (1.534) UBS Securities Australia Ltd Sell 8.560 (1.524) UBS Securities Australia Ltd Sell 8.650 (1.524) UBS Securities Australia Ltd Sell 8.660 (1.066) UBS Securities Australia Ltd Sell 8.753 (6.01) UBS Securities Australia Ltd Sell 8.753 (6.02) UBS Securities Australia Ltd Sell 8.753 (6.02) UBS Securities Australia Ltd Sell 8.753 (6.02) UBS Securit	T	BS Securities Australia Ltd	le l	9//1	5	Ordinary
UBS Securities Australia Ltd Sell 8,922 (402) UBS Securities Australia Ltd Sell 8,922 (402) UBS Securities Australia Ltd Sell 3,655 (402) UBS Securities Australia Ltd Sell 3,655 (402) UBS Securities Australia Ltd Sell 3,655 (402) UBS Securities Australia Ltd Sell 1,106 (41) UBS Securities Australia Ltd Sell 1,016 (11) UBS Securities Australia Ltd Sell 8,598 (42) UBS Securities Australia Ltd Sell 8,598 (42) UBS Securities Australia Ltd Sell 8,598 (42) UBS Securities Australia Ltd Sell 8,455 (50) UBS Securities Australia Ltd Sell 8,455 (50) UBS Securities Australia Ltd Sell 8,455 (50) UBS Securities Australia Ltd Sell 8,435 (50) UBS Securities Australia Ltd Sell 8,435 (60) UBS Securities Australia Ltd	+	bs securities Australia Ltd	lie i	967		Ordinary
UBS Securities Australia Ltd Sell 8.98 (1.09) UBS Securities Australia Ltd Sell 3.610 (390) UBS Securities Australia Ltd Sell 3.650 (4.03) UBS Securities Australia Ltd Sell 1.016 (1.12) UBS Securities Australia Ltd Sell 1.016 (1.12) UBS Securities Australia Ltd Sell 1.016 (1.055) UBS Securities Australia Ltd Sell 8.928 (1.640) UBS Securities Australia Ltd Sell 8.928 (1.651) UBS Securities Australia Ltd Sell 8.928 (1.651) UBS Securities Australia Ltd Sell 8.928 (1.690) UBS Securities Australia Ltd Sell 8.928 (1.600) UBS Securities Australia Ltd Sell 8.928 (1.600) UBS S	7	BS Securities Australia Ltd	Sell	7,810		Ordinary
UBS Securities Australia Ltd Sell 3,552 (403) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 118 (1,13) UBS Securities Australia Ltd Sell 1,000 (4,21) UBS Securities Australia Ltd Sell 1,000 (4,21) UBS Securities Australia Ltd Sell 1,000 (4,21) UBS Securities Australia Ltd Sell 1,000 (4,25) UBS Securities Australia Ltd Sell 8,598 (4,40) UBS Securities Australia Ltd Sell 8,159 (4,50) UBS Securities Australia Ltd Sell 8,150 (5,50) UBS Securities Australia Ltd Sell 8,150 (7,50) UBS Securities Australia Ltd Sell 8,250 (5,00) UBS Securities Australia Ltd Sell 8,344 (9,20) UBS Securities Australia Ltd Sell 8,344 (9,20) UBS Securities Australia Ltd Sell 8,344 (9,20) UBS Securities Au	Ť	BS Securities Australia Ltd	Jan Jan Jan Jan Jan Jan Jan Jan Jan Jan	886		Ordinary
UBS Securities Australia Ltd Sell 3,510 (403) UBS Securities Australia Ltd Sell 3,510 (403) UBS Securities Australia Ltd Sell 1,016 (1,513) UBS Securities Australia Ltd Sell 1,016 (1,23) UBS Securities Australia Ltd Sell 1,016 (1,23) UBS Securities Australia Ltd Sell 1,016 (1,23) UBS Securities Australia Ltd Sell 8,598 (1,402) UBS Securities Australia Ltd Sell 8,106 (1,503) UBS Securities Australia Ltd Sell 8,106 (1,504) UBS Securities Australia Ltd Sell 8,106 (1,509) UBS Securities Australia Ltd Sell 8,100 (1,07) UBS Securities Australia Ltd Sell 8,100 (1,07) UBS Securities Australia Ltd Sell 8,235 (500) UBS Securities Australia Ltd Sell 8,235 (500) UBS Securities Australia Ltd Sell 8,235 (500) UBS Securitie	Ť	85 Securities Australia Ltd	Jell V	8,362		Ordinary
UBS Securities Australia Ltd Sell 1,555 (403) UBS Securities Australia Ltd Sell 13,732 (1,513) UBS Securities Australia Ltd Sell 1,16 (1,12) UBS Securities Australia Ltd Sell 1,06 (1,12) UBS Securities Australia Ltd Sell 8,596 (5,69) UBS Securities Australia Ltd Sell 8,596 (1,055) UBS Securities Australia Ltd Sell 8,596 (1,057) UBS Securities Australia Ltd Sell 8,596 (5,00) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd Sell 8,434 (500) UBS Securities Australia Ltd Sell 8,434 (500) UBS Securities Australia Ltd Sell 8,434 (500) UBS Securitie	┱	85 Securities Australia Ltd	l sell	3,610		Ordinary
UBS Securities Australia Ltd Sell 13,723 (1,131) UBS Securities Australia Ltd Sell 1,016 (1,12) UBS Securities Australia Ltd Sell 1,016 (1,05) UBS Securities Australia Ltd Sell 1,016 (1,05) UBS Securities Australia Ltd Sell 8,596 (1,05) UBS Securities Australia Ltd Sell 8,596 (1,05) UBS Securities Australia Ltd Sell 8,165 (1,09) UBS Securities Australia Ltd Sell 8,165 (1,09) UBS Securities Australia Ltd Sell 8,165 (1,09) UBS Securities Australia Ltd Sell 8,165 (1,00) UBS Securities Australia Ltd Sell 8,100 (1,07) UBS Securities Australia Ltd Sell 8,310 (1,07) UBS Securities Australia Ltd Sell 8,310 (1,07) UBS Securities Australia Ltd Sell 8,34 (9,20) UBS Securities Australia Ltd Sell 8,30 (1,07) UBS Securitie	7	85 Securities Australia Ltd	Jan 1	3,655		
UBS Securities Australia Ltd Sell 1,016 (1,23) UBS Securities Australia Ltd Sell 1,016 (1,13) UBS Securities Australia Ltd Sell 1,016 (1,13) UBS Securities Australia Ltd Sell 8,596 (1,065) UBS Securities Australia Ltd Sell 8,598 (598) UBS Securities Australia Ltd Sell 8,593 (61) UBS Securities Australia Ltd Sell 8,593 (61) UBS Securities Australia Ltd Sell 8,435 (50) UBS Securities Australia Ltd </td <td>╅</td> <td>55 Securities Australia Ltd</td> <td>Ne.</td> <td>13,/23</td> <td></td> <td> </td>	╅	55 Securities Australia Ltd	Ne.	13,/23		
UBS Securities Australia Ltd Sell 1,016 (112) UBS Securities Australia Ltd Sell 461 (53) UBS Securities Australia Ltd Sell 461 (53) UBS Securities Australia Ltd Sell 8598 (48) UBS Securities Australia Ltd Sell 879 (55) UBS Securities Australia Ltd Sell 879 (55) UBS Securities Australia Ltd Sell 870 (73) UBS Securities Australia Ltd Sell 8,345 (50) UBS Securities Australia Ltd Sell 8,345 (50) UBS Securities Australia Ltd Sell 8,345 (61) UBS Securities Australia Ltd Sell 8,344 (92) UBS Securities Australia Ltd Sell 8,345 (61) UBS Securities Australia Ltd Sell	7	85 Securities Australia Ltd	Sell	20,326		1
UBS Securities Australia Ltd Sell 4.01 (17) UBS Securities Australia Ltd Sell 460 (1055) UBS Securities Australia Ltd Sell 460 (1055) UBS Securities Australia Ltd Sell 560 (1055) UBS Securities Australia Ltd Sell 572 (159) UBS Securities Australia Ltd Sell 572 (159) UBS Securities Australia Ltd Sell 572 (159) UBS Securities Australia Ltd Sell 581 (590) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd	+	bs securities Australia Ltd	Jiec	118	(13)	Ordinary
UBS Securities Australia Ltd Sell 481 (55) UBS Securities Australia Ltd Sell 960 (1055) UBS Securities Australia Ltd Sell 815 (90) UBS Securities Australia Ltd Sell 2,304 (254) UBS Securities Australia Ltd Sell 8,598 (98) UBS Securities Australia Ltd Sell 8,163 (90) UBS Securities Australia Ltd Sell 8,163 (90) UBS Securities Australia Ltd Sell 8,163 (90) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd Sell 8,525 (500) UBS Securities Australia Ltd	+	85 Securities Australia Ltd	liec .	1,016	(711)	Ordinary
UBS Securities Australia Ltd Sell 8159 (1,05) UBS Securities Australia Ltd Sell 8,598 (948) UBS Securities Australia Ltd Sell 8,598 (948) UBS Securities Australia Ltd Sell 8,753 (19) UBS Securities Australia Ltd Sell 8,163 (900) UBS Securities Australia Ltd Sell 8,163 (900) UBS Securities Australia Ltd Sell 8,163 (900) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,562 (730) UBS Securities Australia Ltd Sell 8,766 (730) UBS Securities Australia Ltd	۲	bo securities Australia de Lita	iac i	481	(60)	Ordinary
UBS Securities Australia Ltd Sell 815 (40) UBS Securities Australia Ltd Sell 8,598 (348) UBS Securities Australia Ltd Sell 2,304 (754) UBS Securities Australia Ltd Sell 6,467 (713) UBS Securities Australia Ltd Sell 8,698 (368) UBS Securities Australia Ltd Sell 8,105 (300) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,100 (159) UBS Securities Australia Ltd Sell 8,234 (500) UBS Securities Australia Ltd Sell 8,344 (92) UBS Securities Australia Ltd Sell 8,434 (92) UBS Securities Australia Ltd	+	bs securities Australia Ltd	Sell	099'6	(1,065)	Ordinary
UBS Securities Australia Ltd Sell 8,598 (548) UBS Securities Australia Ltd Sell 2,304 (73) UBS Securities Australia Ltd Sell 1,72 (19) UBS Securities Australia Ltd Sell 1,72 (19) UBS Securities Australia Ltd Sell 2,27 (52) UBS Securities Australia Ltd Sell 2,27 (52) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,52 (43) UBS Securities Australia Ltd Sell 1,536 (1,69) UBS Securities Australia Ltd	_	85 Securities Australia Ltd	llec.	816	(06)	Ordinary
UBS Securities Australia Ltd Sell 5.304 (254) UBS Securities Australia Ltd Sell 5.90 (56) UBS Securities Australia Ltd Sell 8.75 (7.13) UBS Securities Australia Ltd Sell 8.75 (7.13) UBS Securities Australia Ltd Sell 4.535 (500) UBS Securities Australia Ltd Sell 4.535 (500) UBS Securities Australia Ltd Sell 4.535 (500) UBS Securities Australia Ltd Sell 8.435 (500) UBS Securities Australia Ltd Sell 8.435 (500) UBS Securities Australia Ltd Sell 8.435 (60) UBS Securities Australia Ltd Sell 8.435 (60) UBS Securities Australia Ltd Sell 8.435 (60) UBS Securities Australia Ltd Sell 8.435 (61) UBS Securities Australia Ltd Sell 8.444 (62) UBS Securities Australia Ltd Sell 8.244 (73) UBS Securities Australia Ltd	1	35 Securities Australia Ltd	Sell	8,598	(948)	Ordinary
UBS Securities Australia Ltd Sell 6,457 (1913) UBS Securities Australia Ltd Sell 6,457 (1913) UBS Securities Australia Ltd Sell 875 (193) UBS Securities Australia Ltd Sell 8,153 (500) UBS Securities Australia Ltd Sell 14,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,100 (823) UBS Securities Australia Ltd Sell 8,100 (823) UBS Securities Australia Ltd Sell 8,344 (920) UBS Securities Australia Ltd Sell 8,344 (920) UBS Securities Australia Ltd Sell 8,344 (920) UBS Securities Australia Ltd Sell 8,562 (500) UBS Securities Australia Ltd Sell 8,344 (920) UBS Securities Australia Ltd Sell 1,259 (1,101) UBS Securities Australia Ltd Sell 1,269 (1,209) UBS Securities Australi	7	35 Securities Australia Ltd	Sell	2,304	(254)	Ordinary
UBS Securities Australia Ltd Sell 6,467 (713) UBS Securities Australia Ltd Sell 172 (19) UBS Securities Australia Ltd Sell 8,163 (500) UBS Securities Australia Ltd Sell 8,163 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,543 (61) UBS Securities Australia Ltd Sell 8,562 (44) UBS Securities Australia Ltd Sell 1,6916 (1,209) UBS Securities Australia Ltd Sell 1,6916 (1,200) UBS Securities Australia Ltd </td <td>٦</td> <td>3S Securities Australia Ltd</td> <td>Selí</td> <td>508</td> <td>(29)</td> <td>Ordinary</td>	٦	3S Securities Australia Ltd	Selí	508	(29)	Ordinary
UBS Securities Australia Ltd Sell 172 (19) UBS Securities Australia Ltd Sell 8.753 (90) UBS Securities Australia Ltd Sell 8.753 (90) UBS Securities Australia Ltd Sell 4.535 (500) UBS Securities Australia Ltd Sell 4.535 (500) UBS Securities Australia Ltd Sell 8.435 (50) UBS Securities Australia Ltd Sell 8.435 (50) UBS Securities Australia Ltd Sell 8.435 (50) UBS Securities Australia Ltd Sell 8.435 (61) UBS Securities Australia Ltd Sell 8.435 (62) UBS Securities Australia Ltd Sell 8.744 (82) UBS Securities Australia Ltd Sell 8.756 (64) UBS Securities Australia Ltd Sell 8.76 (71) UBS Securities Australia Ltd Sell 8.71 (72) UBS Securities Australia Ltd Sell 1.67 (1.640) UBS Securities Australia Ltd <	_	3S Securities Australia Ltd	Sell	6,467	(213)	Ordinary
UBS Securities Australia Ltd Sell R89 (98) UBS Securities Australia Ltd Sell 8,163 (90) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,435 (60) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,344 (92) UBS Securities Australia Ltd Sell 8,344 (92) UBS Securities Australia Ltd Sell 8,522 (54) UBS Securities Australia Ltd Sell 8,522 (54) UBS Securities Australia Ltd Sell 8,522 (54) UBS Securities Australia Ltd Sell 1,675 (10) UBS Securities Australia Ltd Sell 1,571 (1,40) UBS Securities Australia Ltd	-	3S Securities Australia Ltd	Sell	172	(61)	Ordinary
UBS Securities Australia Ltd Sell 8,163 (900) UBS Securities Australia Ltd Sell 4,533 (500) UBS Securities Australia Ltd Sell 14,503 (1,599) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd Sell 8,435 (500) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,435 (61) UBS Securities Australia Ltd Sell 8,344 (82) UBS Securities Australia Ltd Sell 8,344 (82) UBS Securities Australia Ltd Sell 8,562 (61) UBS Securities Australia Ltd Sell 8,562 (64) UBS Securities Australia Ltd Sell 8,562 (107) UBS Securities Australia Ltd Sell 1,6,916 (1,209) UBS Securities Australia Ltd Sell 1,6,916 (1,440) UBS Securities Australia Ltd Sell 1,6,916 (1,440) UBS Securities Aust	-	3S Securities Australia Ltd	Sell	688	(86)	Ordinary
UBS Securities Australia Ltd Sell 227 (25) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,53 (61) UBS Securities Australia Ltd Sell 8,34 (92) UBS Securities Australia Ltd Sell 8,344 (92) UBS Securities Australia Ltd Sell 8,344 (92) UBS Securities Australia Ltd Sell 8,344 (92) UBS Securities Australia Ltd Sell 8,562 (61) UBS Securities Australia Ltd Sell 8,562 (64) UBS Securities Australia Ltd Sell 8,562 (64) UBS Securities Australia Ltd Sell 8,562 (64) UBS Securities Australia Ltd Sell 1,6916 (1,209) UBS Securities Australia Ltd Sell 1,6916 (1,209) UBS Securities Australia Ltd Sell 1,6916 (1,209) UBS Securities Australia Ltd	۲	35 Securities Australia Ltd	Sell	8,163	(006)	Ordinary
UBS Securities Australia Ltd Sell 14,535 (500) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 4,535 (600) UBS Securities Australia Ltd Sell 8,435 (930) UBS Securities Australia Ltd Sell 8,435 (630) UBS Securities Australia Ltd Sell 8,435 (63) UBS Securities Australia Ltd Sell 8,34 (82) UBS Securities Australia Ltd Sell 8,34 (82) UBS Securities Australia Ltd Sell 8,44 (92) UBS Securities Australia Ltd Sell 8,552 (44) UBS Securities Australia Ltd Sell 8,562 (43) UBS Securities Australia Ltd Sell 1,596 (1,209) UBS Securities Australia Ltd Sell 1,596 (1,209) UBS Securities Australia Ltd Sell 1,577 (1,386) UBS Securities Australia Ltd Sell 1,0,366 (1,209) UBS Securities Australia L	T	3S Securities Australia Ltd	Sell	227	(25)	Ordinary
UBS Securities Australia Ltd Sell 14,503 (1,599) UBS Securities Australia Ltd Sell 4,535 (500) UBS Securities Australia Ltd Sell 8,435 (610) UBS Securities Australia Ltd Sell 8,100 (822) UBS Securities Australia Ltd Sell 8,100 (822) UBS Securities Australia Ltd Sell 8,344 (82) UBS Securities Australia Ltd Sell 8,344 (82) UBS Securities Australia Ltd Sell 8,344 (107) UBS Securities Australia Ltd Sell 8,562 (344) UBS Securities Australia Ltd Sell 8,562 (344) UBS Securities Australia Ltd Sell 10,966 (1,209) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities	Ť	3S Securities Australia Etd	Sell	4.535	(200)	Ordinary
UBS Securities Australia Ltd Seil 4,535 (1,520) UBS Securities Australia Ltd Seil 8,35 (61) UBS Securities Australia Ltd Seil 8,34 (92) UBS Securities Australia Ltd Seil 8,34 (82) UBS Securities Australia Ltd Seil 7,44 (82) UBS Securities Australia Ltd Seil 8,344 (92) UBS Securities Australia Ltd Seil 8,344 (92) UBS Securities Australia Ltd Seil 8,562 (64) UBS Securities Australia Ltd Seil 8,562 (58) UBS Securities Australia Ltd Seil 10,966 (1,209) UBS Securities Australia Ltd Seil 10,966 (1,209) UBS Securities Australia Ltd Seil 10,966 (1,209) UBS Securities Australia Ltd Seil 1,676 (1,00) UBS Securities Australia Ltd Seil 1,676 (1,00) UBS Securities Australia Ltd Seil 1,676 (1,00) UBS Securities Australia L	t	35 Sequities Australia Itd	Cell	14 503	(003 1/	Ordinan
UBS Securities Australia Ltd Sell 4,253 (50) UBS Securities Australia Ltd Sell 8,435 (90) UBS Securities Australia Ltd Sell 8,100 (83) UBS Securities Australia Ltd Sell 7,44 (82) UBS Securities Australia Ltd Sell 8,344 (92) UBS Securities Australia Ltd Sell 8,44 (93) UBS Securities Australia Ltd Sell 8,552 (44) UBS Securities Australia Ltd Sell 8,552 (44) UBS Securities Australia Ltd Sell (1,59) (1,59) UBS Securities Australia Ltd Sell (1,59) (1,50) UBS Securities Australia Ltd Sell (1,69) (1,690) UBS Securities Australia Ltd Sell (1,690) (1,690) UBS Securities	1	O Committee Application Line	100	200,41	(660')	Ordinary
UBS Securities Australia Ltd Sell 8.435 (91) UBS Securities Australia Ltd Sell 8.435 (92) UBS Securities Australia Ltd Sell 8.344 (82) UBS Securities Australia Ltd Sell 7.44 (82) UBS Securities Australia Ltd Sell 8.344 (92) UBS Securities Australia Ltd Sell 8.752 (61) UBS Securities Australia Ltd Sell 8.752 (93) UBS Securities Australia Ltd Sell 8.752 (94) UBS Securities Australia Ltd Sell 16,916 (1,07) UBS Securities Australia Ltd Sell 16,916 (1,865) UBS Securities Australia Ltd Sell 1,877 (1,209) UBS Securities Australia Ltd Sell 1,877 (1,40) UBS Securities Australia Ltd Sell 1,877 (1,40) UBS Securities Australia Ltd Sell 1,674 (1,14) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia L	Ť	20 Commission Americally 14d	170	633	(100)	Ordinary
UBS Securities Australia Ltd Sell 8,435 (92) UBS Securities Australia Ltd Sell 8,100 (893) UBS Securities Australia Ltd Sell 8,100 (893) UBS Securities Australia Ltd Sell 8,344 (82) UBS Securities Australia Ltd Sell 8,344 (920) UBS Securities Australia Ltd Sell 8,444 (920) UBS Securities Australia Ltd Sell 8,562 (944) UBS Securities Australia Ltd Sell 8,562 (944) UBS Securities Australia Ltd Sell 16,966 (1,209) UBS Securities Australia Ltd Sell 16,966 (1,209) UBS Securities Australia Ltd Sell 1,577 (1,386) UBS Securities Australia Ltd Sell 1,577 (1,386) UBS Securities Australia Ltd Sell 1,577 (1,40) UBS Securities Australia Ltd Sell 2,131 (30) UBS Securities Australia Ltd Sell 2,131 (30) UBS Securities Austra	Ť	oo securities Australia Liu	ilac.	200	(10)	Ordinary
UBS Securities Australia Ltd Sell 834 (82) UBS Securities Australia Ltd Sell 744 (82) UBS Securities Australia Ltd Sell 744 (82) UBS Securities Australia Ltd Sell 870 (107) UBS Securities Australia Ltd Sell 874 (920) UBS Securities Australia Ltd Sell 876 (107) UBS Securities Australia Ltd Sell 16,916 (107) UBS Securities Australia Ltd Sell 16,916 (1,209) UBS Securities Australia Ltd Sell 1,036 (1,40) UBS Securities Australia Ltd Sell 1,036 (1,40) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 2,131 (255) UBS Securities Australia Ltd </td <td>Ť</td> <td>SS Securities Australia Liu</td> <td>II-U</td> <td>6,435</td> <td>(056)</td> <td>Orginary</td>	Ť	SS Securities Australia Liu	II-U	6,435	(056)	Orginary
UBS Securities Australia Ltd Sell 8,100 (82) UBS Securities Australia Ltd Sell 553 (61) UBS Securities Australia Ltd Sell 8,244 (92) UBS Securities Australia Ltd Sell 844 (93) UBS Securities Australia Ltd Sell 8,562 (64) UBS Securities Australia Ltd Sell 8,562 (64) UBS Securities Australia Ltd Sell 16,916 (17) UBS Securities Australia Ltd Sell 1,671 (17) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,670 (1,40) UBS Securities Australia Ltd Sell 1,674 (1,40) UBS Securities Australia Ltd Sell 1,630 (1,40) UBS Securities Australia Ltd Sell 2,04 (255) UBS Securities Australia Ltd	- 1	SS Securities Australia LIO	liac	834	(76)	Orginary
UBS Securities Australia Ltd Sell 744 (82) UBS Securities Australia Ltd Sell 534 (62) UBS Securities Australia Ltd Sell 8,344 (920) UBS Securities Australia Ltd Sell 8,444 (920) UBS Securities Australia Ltd Sell 8,562 (944) UBS Securities Australia Ltd Sell 8,562 (943) UBS Securities Australia Ltd Sell 16,966 (1,209) UBS Securities Australia Ltd Sell 1,596 (1,209) UBS Securities Australia Ltd Sell 1,596 (1,209) UBS Securities Australia Ltd Sell 1,2,577 (1,366) UBS Securities Australia Ltd Sell 1,324 (1,40) UBS Securities Australia Ltd Sell 2,31 (30) UBS Securities Australia Ltd Sell 2,31 (30) UBS Securities Australia Ltd Sell 2,404 (1,40) UBS Securities Australia Ltd Sell 2,404 (7,16) UBS Securities Australia	- 1	35 Securities Australia Ltd	Sell	8,100	(863)	Ordinary
UBS Securities Australia Ltd Sell 653 (61) UBS Securities Australia Ltd Sell 8344 (920) UBS Securities Australia Ltd Sell 844 (93) UBS Securities Australia Ltd Sell 8,562 (944) UBS Securities Australia Ltd Sell 5,562 (58) UBS Securities Australia Ltd Sell 16,916 (1,209) UBS Securities Australia Ltd Sell 10,966 (1,209) UBS Securities Australia Ltd Sell 1,877 (209) UBS Securities Australia Ltd Sell 1,677 (1,209) UBS Securities Australia Ltd Sell 1,677 (1,209) UBS Securities Australia Ltd Sell 1,677 (1,40) UBS Securities Australia Ltd Sell 1,634 (1,14) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 2,404 (255) UBS Securities Austral		3S Securities Australia Ltd	Sell	744	(82)	Ordinary
UBS Securities Australia Ltd Sell 8.344 (920) UBS Securities Australia Ltd Sell 870 (107) UBS Securities Australia Ltd Sell 8,562 (944) UBS Securities Australia Ltd Sell 526 (944) UBS Securities Australia Ltd Sell 16,916 (1,865) UBS Securities Australia Ltd Sell 16,916 (1,865) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,5,71 (1,386) UBS Securities Australia Ltd Sell 1,5,71 (1,386) UBS Securities Australia Ltd Sell 2,131 (30) UBS Securities Australia Ltd Sell 2,131 (40) UBS Securities Australia Ltd Sell 2,131 (40) UBS Securities Australia Ltd Sell 2,141 (716) UBS Securities Australia Ltd Sell 2,744 (716) UBS Securities Australi		35 Securities Australia Ltd	Sell	223	(19)	Ordinary
UBS Securities Australia Ltd Sell 970 (107) UBS Securities Australia Ltd Sell 844 (83) UBS Securities Australia Ltd Sell 8,562 (944) UBS Securities Australia Ltd Sell 526 (58) UBS Securities Australia Ltd Sell 16,916 (1,209) UBS Securities Australia Ltd Sell 16,916 (1,209) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,2,571 (1,386) UBS Securities Australia Ltd Sell 1,2,571 (1,40) UBS Securities Australia Ltd Sell 1,034 (1,40) UBS Securities Australia Ltd Sell 2,131 (2,35) UBS Securities Australia Ltd Sell 2,131 (3,60) UBS Securities Australia Ltd Sell 2,404 (1,65) UBS Securities Australia Ltd Sell 2,404 (7,66) UBS Securities Australia Ltd Sell 2,404 (7,16) UBS Securities Au		3S Securities Australia Ltd	lleS	8,344	(026)	Ordinary
UBS Securities Australia Ltd Sell 8444 (93) UBS Securities Australia Ltd Sell 5.562 (944) UBS Securities Australia Ltd Sell 5.56 (58) UBS Securities Australia Ltd Sell 16,916 (1,109) UBS Securities Australia Ltd Sell 10,966 (1,209) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,877 (1,840) UBS Securities Australia Ltd Sell 1,689 (1,840) UBS Securities Australia Ltd Sell 1,689 (1,40) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 8,714 (1,696) UBS Securities Australia Ltd Sell 2,04 (2,65) UBS Securities Australia Ltd Sell 2,71 (3) UBS Securities Aust	F	35 Securities Australia Ltd	ll ex	920	(107)	Ordinary
UBS Securities Australia Ltd Sell 8,562 (944) UBS Securities Australia Ltd Sell 526 (58) UBS Securities Australia Ltd Sell 16,916 (1,829) UBS Securities Australia Ltd Sell 10,966 (1,209) UBS Securities Australia Ltd Sell 10,966 (1,209) UBS Securities Australia Ltd Sell 12,571 (1,389) UBS Securities Australia Ltd Sell 2,131 (30) UBS Securities Australia Ltd Sell 2,131 (40) UBS Securities Australia Ltd Sell 2,404 (14) UBS Securities Australia Ltd Sell 2,404 (16) UBS Securities Australia Ltd Sell 2,404 (16) UBS Securities Australia Ltd Sell 2,704 (716) UBS Securities Australia	f	3S Securities Australia I td	Sett	844	(85)	Ordinary
UBS Securities Australia Ltd Sell CFD UBS Securities Australia Ltd Sell 16,916 (1) UBS Securities Australia Ltd Sell 16,916 (1,855) UBS Securities Australia Ltd Sell 10,966 (1,209) UBS Securities Australia Ltd Sell 1,877 (209) UBS Securities Australia Ltd Sell 1,577 (1,209) UBS Securities Australia Ltd Sell 1,587 (1,840) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 2,131 (255) UBS Securities Australia Ltd Sell 2,404 (265) UBS Securities Australia Ltd Sell 8,714 (96) UBS Securities Australia Ltd Sell 2,704 (716) UBS Securities Australia Ltd Sell 2,704 (716) UBS Securities Australia Ltd Sell (716) (716) UBS Securities Australia Ltd <		35 Securities Australia Itd	Z S	8 562	(VVO)	Ordinary
UBS Securities Australia Ltd Sell 15,916 (15) UBS Securities Australia Ltd Sell 16,916 (1,865) UBS Securities Australia Ltd Sell 16,916 (1,209) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,877 (1,386) UBS Securities Australia Ltd Sell 1,674 (1,40) UBS Securities Australia Ltd Sell 2,131 (39) UBS Securities Australia Ltd Sell 2,404 (1,14) UBS Securities Australia Ltd Sell 2,404 (1,16) UBS Securities Australia Ltd Sell 2,704 (1,16) UBS Securities Australia Ltd Sell 2,704 (7,16) UBS Securities Australia Ltd Sell 2,704 (7,16) UBS Securities Australia Ltd Sell 2,704 (7,16) UBS Securities Australia Ltd Sell (2,94 (7,16) UBS Securitie	1	SC Societies Australia Ltd	100	200,0	(446)	Cidinary
UBS Securities Australia Ltd Sell 16,916 (1,620) UBS Securities Australia Ltd Sell 16,916 (1,620) UBS Securities Australia Ltd Sell 1,77 (27) UBS Securities Australia Ltd Sell 1,877 (27) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,689 (1,840) UBS Securities Australia Ltd Sell 2,131 (235) UBS Securities Australia Ltd Sell 363 (40) UBS Securities Australia Ltd Sell 2,404 (14) UBS Securities Australia Ltd Sell 871 (40) UBS Securities Australia Ltd Sell 2,404 (265) UBS Securities Australia Ltd Sell 871 (40) UBS Securities Australia Ltd Sell 2,7 (716) UBS Securities Australia Ltd Sell 27 (716) UBS Securities Australia Ltd Sell (874 (716) UBS Securities Australia Ltd	1	So securities Australia Ltd	Jac Jac Jac Jac Jac Jac Jac Jac Jac Jac	975	(85)	Orginary
UBS Securities Australia Ltd Sell (1,855) UBS Securities Australia Ltd Sell 10,966 (1,209) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,877 (1,840) UBS Securities Australia Ltd Sell 1,571 (1,840) UBS Securities Australia Ltd Sell 1,689 (1,840) UBS Securities Australia Ltd Sell 363 (40) UBS Securities Australia Ltd Sell 1,034 (114) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 2,704 (716) UBS Securities Australia Ltd Sell 2,71 (95) UBS Securities Australia Ltd Sell 2,71 (96) UBS Securities Australia Ltd Sell 2,71 (716) UBS Securities Australia Ltd Sell 8,716 (96) UBS Securities Australia Ltd Sell	- 1	ss securities Australia Ltd	Sell	6	(E)	Ordinary
UBS Securities Australia Ltd Sell 10,966 (1,209) UBS Securities Australia Ltd Sell 1377 (207) UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,577 (1,386) UBS Securities Australia Ltd Sell 2,131 (385) UBS Securities Australia Ltd Sell 2,131 (407) UBS Securities Australia Ltd Sell 2,404 (114) UBS Securities Australia Ltd Sell 2,404 (36) UBS Securities Australia Ltd Sell 2,704 (716) UBS Securities Australia Ltd Sell 2,71 (716) UBS Securities Australia Ltd Sell 2,71 (716) UBS Securities Australia Ltd Sell (8,94 (716) UBS Securities Australia Lt	7	35 Securities Australia Ltd	Sell	16,916	(1,865)	Ordinary
UBS Securities Australia Ltd Sell 1,77 (57) UBS Securities Australia Ltd Sell 1,877 (707) UBS Securities Australia Ltd Sell 1,5,571 (1,380) UBS Securities Australia Ltd Sell 1,6,840 (1,840) UBS Securities Australia Ltd Sell 2,131 (2,35) UBS Securities Australia Ltd Sell 1,034 (1,14) UBS Securities Australia Ltd Sell 2,404 (265) UBS Securities Australia Ltd Sell 8,704 (765) UBS Securities Australia Ltd Sell 2,704 (765) UBS Securities Australia Ltd Sell 2,704 (716) UBS Securities Australia Ltd Sell 2,704 (716) UBS Securities Australia Ltd Sell 7,716 (716) UBS Securities Australia Ltd Sell (716) (716) UBS Securities Australia Ltd Sell (716) (716) UBS Securities Australia Ltd Sell (716) (716) UBS Securities Austr		3S Securities Australia Ltd	Sell	10,966	(1,209)	Ordinary
UBS Securities Australia Ltd Sell 1,877 (207) UBS Securities Australia Ltd Sell 1,5571 (1,386) UBS Securities Australia Ltd Sell 1,689 (1,440) UBS Securities Australia Ltd Sell 2,131 (235) UBS Securities Australia Ltd Sell 1,034 (114) UBS Securities Australia Ltd Sell 2,404 (715) UBS Securities Australia Ltd Sell 871 (66) UBS Securities Australia Ltd Sell 871 (66) UBS Securities Australia Ltd Sell 871 (716)		3S Securities Australia Ltd	Sell	517	(23)	Ordinary
UBS Securities Australia Ltd Sell 12,571 (1,386) UBS Securities Australia Ltd Sell 1,6889 (1,840) UBS Securities Australia Ltd Sell 2,131 (235) UBS Securities Australia Ltd Sell 363 (40) UBS Securities Australia Ltd Sell 2,404 (114) UBS Securities Australia Ltd Sell 8,71 (36) UBS Securities Australia Ltd Sell 27 (716) UBS Securities Australia Ltd Sell 27 (716) UBS Securities Australia Ltd Sell (716) (716) UBS Securities Australia Ltd Sell (716) (716) UBS Securities Australia Ltd Sell (716) (716)	Ī	35 Securities Australia Ltd	Sell	1,877	(207)	Ordinary
UBS Securities Australia Ltd Sell 16,689 (1,840) UBS Securities Australia Ltd Sell 2,131 (235) UBS Securities Australia Ltd Sell 1,034 (114) UBS Securities Australia Ltd Sell 2,404 (265) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 2,71 (31) UBS Securities Australia Ltd Sell 181 (716) UBS Securities Australia Ltd Sell (876) (8716) UBS Securities Australia Ltd Sell (8716) (8716)	Ē	3S Securities Australia Ltd	Sell	12,571	(1,386)	Ordinary
UBS Securities Australia Ltd Sell 2,131 (235) UBS Securities Australia Ltd Sell 363 (40) UBS Securities Australia Ltd Sell 2,404 (714) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 6,794 (716) UBS Securities Australia Ltd Sell 6,716 (961) UBS Securities Australia Ltd Sell 8,716 (961)	┪	35 Securities Australia Ltd	Sell	16.689	(1.840)	Ordinary
UBS Securities Australia Ltd Sell 363 (40) UBS Securities Australia Ltd Sell 1,034 (114) UBS Securities Australia Ltd Sell 2,404 (25) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 27 (7) UBS Securities Australia Ltd Sell (7 16)	Ť	35 Securities Australia Ltd	Seli	2 131	(235)	Ordinary
UBS Securities Australia Ltd Sell 1,034 (114) UBS Securities Australia Ltd Sell 2,404 (265) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 2,7 (76) UBS Securities Australia Ltd Sell 2,7 (76) UBS Securities Australia Ltd Sell 181 (76) UBS Securities Australia Ltd Sell 181 (76) UBS Securities Australia Ltd Sell 8,716 (961)	T	S Securities Australia 14d	3	363	(007)	Ordinary
UBS Securities Australia Ltd Sell 2,404 (265) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 871 (96) UBS Securities Australia Ltd Sell 6,494 (716) UBS Securities Australia Ltd Sell 6,494 (716) UBS Securities Australia Ltd Sell 8,716 (96)	+	S Societies Australia 14d	3	1 034	(414)	Ordinary
UBS Securities Australia Ltd Sell 2,704 (150) UBS Securities Australia Ltd Sell 27 (3) UBS Securities Australia Ltd Sell 27 (7) UBS Securities Australia Ltd Sell (7 lb) UBS Securities Australia Ltd Sell (7 lb) UBS Securities Australia Ltd Sell (8,716) UBS Securities Australia Ltd Sell (8,716)	T	Contribut Australia Led	100	+CO. L	(+11)	Cidinary
UBS Securities Australia Ltd Sell 871 (9b.) UBS Securities Australia Ltd Sell 27 (3) UBS Securities Australia Ltd Sell 6,494 (716) UBS Securities Australia Ltd Sell 6,494 (716) UBS Securities Australia Ltd Sell 181 (20) UBS Securities Australia Ltd Sell 8,716 (961)	+	so securities Australia Liu		2,404	(592)	Ordinary
UBS Securities Australia Ltd Sell 6,494 (716) UBS Securities Australia Ltd Sell 6,494 (716) UBS Securities Australia Ltd Sell 6,494 (716) UBS Securities Australia Ltd Sell 181 (20) UBS Securities Australia Ltd Sell 8,716 (961)	Ť	ss securities Australia Ltd	Sell	8/1	(96)	Ordinary
UBS Securities Australia Ltd Seil 6,494 (716) UBS Securities Australia Ltd Seil 181 (20) UBS Securities Australia Ltd Seil 8,716 (961)	Ť	S Securities Australia Ltd	les.	77	(3)	Ordinary
UBS Securities Australia Ltd Seli 181 (20) UBS Securities Australia Ltd Seli 8,716 (961)	7	is securities Australia Ltd	Sei	6,494	(716)	Ordinary
UBS Securities Australia Ltd Sell (961)	_	iS Securities Australia Ltd	Sell	181	(20)	Ordinary
	Г	S Securities Australia I to	راوا	8716	(961)	Ordinary
	Т	כן שברתן ווובי אחזון מוום דירי	Jac	21,10	(106)	Oralisiy

	II Bo God without Australia (40)	Sel	677	(49)	Ordinary
1 2-May-14	T			(5.1)	
12 May 14	Ups Securities Australia Ltd	la C	7,907	(208)	Ordinary
12 14 14	Ť	Neil Neil	688	(75)	Ordinar
12-May-14	UBS Securities Australia Ltd	Sell	4,749	(219)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sel	915	(100)	Ordinar
12-May-14	UBS Securities Australia Ltd	Sell	1,308	(143)	Ordinary
12-May-14	UBS Securities Australia Lto	Sell	12,810	(1,400)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	8,244	(006)	Ordinar
12-May-14	UBS Securities Australia Ltd	Sell	1.328	(145)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	1.081	(118)	Ordinary
12-May-14	UBS Securities Australia Ltd	Seil	1 246	(136)	Ordinary
12-May-14	UBS Securities Australia Ltd	lies	1 501	(164)	i i
12-May-14	UBS Securities Australia 1td	9	1 713	(107)	i de
12-May-14	I IRS Securities Australia I to	Cell	1000	(787)	Oldinary
12 1/2: 14	Dis Securities Australia Publication	II O	7,001	(784)	Ordinar
12-IVIdy-14	Obs securities Australia Ltd	Self	3,426	(3/4)	Ordinary
12-May-14	UBS Securities Australia Ltd	llex.	17,568	(1,920)	Ordinary
12-May-14	UBS Securities Australia Ltd	Jeli	467	(51)	Ordinary
1 2-May-14	UBS Securities Australia Ltd	38	2,022	(221)	Ordinary
12-May-14	UBS Securities Australia Ltd	les.	4,255	(465)	Ordinar
12-May-14	UBS Securities Australia Ltd	Sell	202	(77)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	866	(109)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	2,544	(278)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	860'8	(882)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	1,848	(202)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	5,042	(551)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	868'1	(207)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	7,311	(662)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	137	(15)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	1,201	(131)	Ordinar
12-May-14	UBS Securities Australia Ltd	Seli	586	(64)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	2.290	(250)	Ordinary
12-May-14	UBS Securities Australia Ltd	es	8	(2)	Ordinan
12-May-14	UBS Securities Australia Ltd	le S	1 933	(711)	Ordinary
12-May-14	LIBS Securities Australia Ltd	اوا	1,072	(11)	Ordinary
12-May-14	I IBS Sportition Australia I to	3	2/0/1	(10401)	
12-May-14	+ IRC Contribute Australia Ltd	100	2,000	(040')	Clouding
12-May-14	1 OC Constitute Australia Ltd	150	1000	(000)	5 6
12 May 14	COO Securities Australia Lite	ii c	0,704	(858)	Ordinary
12 May 14	Libe Committee Australia Ltd	E C	860,1	(021)	Ordinary
12 May 14	Upo Commiss Australia Liu	Jac J	500'5	(370)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	16,799	(1,834)	Ordinary
12-May-14	UBS Securities Australia Ltd	les.	18,320	(2,000)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	10,653	(1,163)	Ordinary
12-May-14	UBS Securities Australia Ltd	Sell	696'61	(2,180)	Ordinary
13-May-14	UBS AG London Branch	lle <u>s</u>	65,942	(7,260)	Ordinary
13-May-14	UBS AG London Branch	Buy	66.204	7.260	Ordinary
13-May-14	UBS Securities Australia Ltd	Şel	132 697	(14 560)	Ordinary
13-May-14	UBS Securities Australia I td	Biry	181 108	10.840	Ordinary June 1
13-Mav-14	ILIBS Securities Australia I td	Sell	56 101	(100.2)	
13-May-14	I IBC Countities Australia I to	Dia.	161,60	1 0041	Cicillary
13-May-14	I IRS Societies Australia 14d	200	100,00	190,1	200
13 May 14	I IDC Cocuration Australia Ltd	100	40,200	002,7	Orginary
13-May-14	URC Securities Australia 14d	Jac G	000 10	(4,354)	Orginar
13-May-14	I BS Securities Australia 14d	Broy	200,10	0,000	Cignialy
12-44-14	1 100 Cocurition Australia Ltd	buy.	201,067	10,000	Orginary
12 14 24 14	Libo Committee Australia Ltd	ng.	901,636	516,75	Ordinary
12 May 14	I DC Cocurition Australia Ltd	a	5/0/16	(10,000)	Orginary
12 May 14	Libs counties Australia Ltd	buy	14,046	חומ'ו	Orginary
13 May 14	Libo comition A retails to	buy P.II.	545,22	2,4/4	Ordinary
12 May 14	LIDO Cocuriètica Australia Lid	buy.	410,17	067'1	Oramary
1 - May 14	Upo Committee Australia Ltd	pay C-ti	1,625,380	200,000	Ordinary
13-May-14	Ubb Securities Australia Ltd	Seil	5,524,641	(604,387)	Ordinary
13-May-14	UBS Global Asset Management	Buy	77,584	8,514	Ordinary
13-May-14	UBS Global Asset Management	Buy	120,203	13,191	Ordinary
13-May-14	UBS Wealth Management Australia Ltd	Buy	383,048	42,000	Ordinary
13-May-14	UBS Wealth Management Australia Ltd	815	273 606	30,000	Ordinan
13-May-14		(200/2/2	000'00	5
1 2-1 YIQY - 1-4	UBS Wealth Management Australia Ltd	Sell	1.481	(163)	Ordinary

1055 Securities Autorisida Ltd 561 562 662 663 6						
Mail Securities Australia Ltd Sell 1,255 (1,150)	12-May-14	TIBS Securities Australia ted	Sell	1/9	(59)	Ordinary
USS Securities Australia Ltd Sell 15,340 1,1500	12-May-14	UBS Securities Australia I td	las	626	(600)	Ordinary
Mail Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australi	12-May-14	LIBS Securities Australia Ltd	liac o	070	(60)	Ordinary
USS Securities Australia Ltd Sell	12 1421. 14	DOD Securities Australia LIO	Jac.	175'01	(1,100)	Ordinary
USS Securities Australia Ltd Sell 1955 Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securiti	12-May-14	UBS Securities Australia Ltd	Sell	8	(2)	Ordinary
USS Securities Australia Ltd Sell 15,245 (1,020) USS Securities Australia Ltd Sell 15,240 (1,020) USS Securities Australia Ltd Sell 1,270 (1,020) USS Securities Australia Ltd Sell 2,262 (1,020) USS Securities Australia Ltd Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 2,262 (1,020) USS Securities Australia Ltd Sell Sell 3,262 (1,020) USS Securities Australia Ltd Sell Sell 1,260 (1,020) USS Securities Australia Ltd Sell Sell 1,260 (1,020) USS Securities Australia Ltd Sell Sell 1,260 (1,020) USS Securities Australia Ltd Sell Sell 3,262 (1,020) (1,020) USS Securities Australia Ltd Sell Sell 3,262 (1,020) (1,020) USS Securities Australia Ltd Sell Sell Sell 3,262 (1,020) (1,020) USS Securities Australia Ltd Sell Sell 3,262 (1,020) (1,020) USS Securities Australia Ltd Sell Sell 3,262 (1,020) (1,020) USS Securities Australia Ltd Sell Sell 3,262 (1,020) (1,020) USS Secur	12-May-14	UBS Securities Australia Ltd	Seit	571	(63)	Ordinary
USS Securities Australia Ltd Sell 165 Securities Australia Ltd Sell 165 Securities Australia Ltd Sell Sell 1720 (1709) 175 Securities Australia Ltd Sell Sell 1720 (1701) 175 Securities Australia Ltd Sell Sell 1720 (1701)	12-May-14	UBS Securities Australia Ltd	Sell	245	(27)	Ordinary
URS Securities Australia Ltd Sell 6.57 (769) URS Securities Australia Ltd Sell 1.954 (1,407) URS Securities Australia Ltd Sell 1.954 (1,407) URS Securities Australia Ltd Sell 2.2693 (2,502) URS Securities Australia Ltd Sell 2.2693 (2,502) URS Securities Australia Ltd Sell 2.2693 (4,007) URS Securities Australia Ltd Sell 2.2693 (4,007) URS Securities Australia Ltd Sell 2.2693 (4,007) URS Securities Australia Ltd Sell 2.79 (4,107) URS Securities Australia Ltd Sell 3.79 (4,107) URS	12-May-14	UBS Securities Australia Ltd	Sell	16,380	(1,806)	Ordinary
UBS Securities Australia Ltd Sell 11,954 (17,09) UBS Securities Australia Ltd Sell 1,270 (14,00) UBS Securities Australia Ltd Sell 2,270 (14,00) (14,0	12-May-14	UBS Securities Australia Ltd	Sell	281	(31)	Ordinary
UBS Securities Australia Ltd Sell 11,954 (1,18) UBS Securities Australia Ltd Sell 12,663 (4,007) UBS Securities Australia Ltd Sell 22,693 (4,007) UBS Securities Australia Ltd Sell 22,693 (4,007) UBS Securities Australia Ltd Sell 22,693 (4,007) UBS Securities Australia Ltd Sell 2,108 (4,007) UBS Securities Australia Ltd Sell 2,108 (4,007) UBS Securities Australia Ltd Sell 2,719 (4,007) UBS Securities Australia Ltd Sell 3,719 (4,107) UBS Securities Australia Ltd Sell 3,729 (4,107) UBS Securities Australia Ltd Sell 3,729 (1,000) UBS Securities Australia Ltd Sell 3,224 (1,001)	12-May-14	UBS Securities Australia Ltd	lley.	6,975	(20)	Ordinary
UBS Securities Australia Ltd Sell 1,279 (400) UBS Securities Australia Ltd Sell 3,237 (400) UBS Securities Australia Ltd Sell 2,281 (50) LBS Securities Australia Ltd Sell 2,060 (1,000) UBS Securities Australia Ltd Sell 3,070 (1,100) UBS Securities Australia Ltd Sell 3,070 (1,000) UBS S	12-May-14	UBS Securities Australia Ltd	Sell	11,954	(1,318)	Ordinary
UBS Securities Australia Ltd Sell Sell Sel S	12-May-14	UBS Securities Australia Ltd	Sell	1,270	(140)	Ordinary
UBS Securities Australia Ltd Sell 2,169 1,050	12-May-14	UBS Securities Australia Ltd	Sell	36.343	(4,007)	Ordinary
UBS Securities Australia Ltd Sell 22,699 (2,592) UBS Securities Australia Ltd Sell 2,269 (3,397) UBS Securities Australia Ltd Sell 2,070 (1,202) UBS Securities Australia Ltd Sell 3,79 (4,002) UBS Securities Australia Ltd Sell 3,70 (1,028) UBS Securities Australia Ltd Sell 3,60 (1,028) UBS Securities Australia Ltd Sell 3,60 (1,000) UBS Securities Australia Ltd Sell 3,60 (1,000) UBS Securities Australia Ltd Sell 3,60 (1,000) UBS Securities Australia Ltd Sell 3,00 (1,000) UBS Secur	12-May-14	UBS Securities Australia Ltd	Self	295	(62)	Ordinary
UBS Securities Australia Ltd Sell 2,768 (3.2) UBS Securities Australia Ltd Sell 9,700 (1,000) UBS Securities Australia Ltd Sell 9,000 (1,000) UBS Securities Australia Ltd Sell 1,128 (1,20) UBS Securities Australia Ltd Sell 1,129 (4,0) UBS Securities Australia Ltd Sell 1,202 (4,0) UBS Securities Australia Ltd Sell 1,273 (4,0) UBS Securities Australia Ltd Sell 1,273 (4,0) UBS Securities Australia Ltd Sell 2,262,500 (5,0) UBS Securities Australia Ltd Sell 2,723 (1,0) UBS Securities Australia Ltd Sell 2,725 (4,0) UBS Securities Australia Ltd Sell 3,200 (2,0) UBS Securities Australia Ltd Sell 3,200 (2,0) UBS Securities Australia Ltd Sell 3,600 (1,0) UBS Securities Australia Ltd Sell 3,600 (1,0) UBS Securities Aust	12-Mav-14	ILIBS Securities Australia I to	la C	205 22	(05)	Ordinary
UBS Securities Australials Ltd Sell 2,168 (239) UBS Securities Australials Ltd Sell 40,000 (1,200) UBS Securities Australia Ltd Sell 1,125 (1,20) UBS Securities Australia Ltd Sell 3,719 (4,00) UBS Securities Australia Ltd Sell 3,729 (1,028) UBS Securities Australia Ltd Sell 2,222 (1,028) UBS Securities Australia Ltd Sell 2,723 (1,028) UBS Securities Australia Ltd Sell 2,722 (1,029) UBS Securities Australia Ltd Sell 2,722 (1,029) UBS Securities Australia Ltd Sell 2,722 (1,029) UBS Securities Australia Ltd Sell 3,625 (4,03) UBS Securities Australia Ltd Sell 3,000 (1,000) U	12-May-14	LIRS Securities Australia I to	3	100	(207)	Ordinary
UBS Securities Australia Ltd Sell 9,00 (1,000) UBS Securities Australia Ltd Sell 9,00 (1,000) UBS Securities Australia Ltd Sell 1,128 (124) UBS Securities Australia Ltd Sell 3,719 (40) UBS Securities Australia Ltd Sell 3,719 (41) UBS Securities Australia Ltd Sell 3,713 (40) UBS Securities Australia Ltd Sell 1,473 (40) UBS Securities Australia Ltd Sell 1,473 (40) UBS Securities Australia Ltd Sell 2,722 (1,00) UBS Securities Australia Ltd Sell 2,723 (1,00) UBS Securities Australia Ltd Sell 3,000 (1,00) UBS Securities Australia Ltd Sell 3,000 (1,00) UBS Securities Australia Ltd Sell 3,000 (1,000) UBS Securities Australia Ltd Sell 3,000 (1,000) UBS Securities Australia Ltd Sell 3,000 (1,000) UBS Securities Austra	12-May-14	LIBS Securities Australia 14d	Coll	107	(1.5)	Ordinary
OS Securities Australia Ltd Sell 40,009 UBS Securities Australia Ltd Sell 1,125 (4,20) UBS Securities Australia Ltd Sell 3,719 (4,10) UBS Securities Australia Ltd Sell 3,719 (4,10) UBS Securities Australia Ltd Sell 2,222 (4,01) UBS Securities Australia Ltd Sell 2,222 (4,02) UBS Securities Australia Ltd Sell 2,222 (4,03) UBS Securities Australia Ltd Sell 2,222 (4,03) UBS Securities Australia Ltd Sell 2,222 (4,03) UBS Securities Australia Ltd Sell 2,262,500 (5,50,000) UBS Securities Australia Ltd Sell 2,262,500 (3,50,000) UBS Securities Australia Ltd Sell 2,262,500 (3,50,000) UBS Securities Australia Ltd Sell 3,654 (4,03) UBS Securities Australia Ltd Sell 3,000 (3,000) UBS Securities Australia Ltd Sell 5,010 (1,000) UBS Securit	12 6600 14	LIDS Countries Australia Ltd	3511	0,100	(862)	Crainary
UBS Securities Australia Ltd Sell 1,005 (4,005 UBS Securities Australia Ltd Sell 37.9 (410) UBS Securities Australia Ltd Sell 2.262.500 (250,000) UBS Securities Australia Ltd Sell 2.262.500 (250,000) UBS Securities Australia Ltd Sell 3.655 (403) UBS Securities Australia Ltd Sell 3.655 (404) UBS Securities Australia Ltd Sell 3.675 (404) UBS Securities Au	12 May 14	LIDE COUNTRIES AUSTRAINE LID	ii s	0/0/6	(000,1)	Ordinary
UBS Securities Australia Ltd Sell 1,125 (1,24) UBS Securities Australia Ltd Sell 3719 (410) UBS Securities Australia Ltd Sell 3719 (410) UBS Securities Australia Ltd Sell 2,222 (425) UBS Securities Australia Ltd Sell 2,222 (426) UBS Securities Australia Ltd Sell 2,222 (426) UBS Securities Australia Ltd Sell 2,222 (426) UBS Securities Australia Ltd Sell 2,225 (427) UBS Securities Australia Ltd Sell 3,525 (439) UBS Securities Australia Ltd Sell 3,654 (404) UBS Securities Australia Ltd Sell 3,675 (404) UBS Securities Australia Ltd<	12-Wdy-14	Ups securities Australia Ltd	l Ne	40,089	(4,420)	Ordinary
UBS Securities Australia Ltd Sell 3719 (410) UBS Securities Australia Ltd Sell 3719 (410) UBS Securities Australia Ltd Sell 3719 (410) UBS Securities Australia Ltd Sell 2,222 (245) UBS Securities Australia Ltd Sell 3,773 (153) UBS Securities Australia Ltd Sell 1,473 (153) UBS Securities Australia Ltd Sell 2,262,500 (250,000) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 3,656 (403) UBS Securities Australia Ltd Sell 3,656 (403) UBS Securities Australia Ltd Sell 3,656 (403) UBS Securities Australi	12-May-14	U65 Securities Australia Ltd	Ne:	1,125	(124)	Ordinary
UBS Securities Australia Ltd Sell 3,719 (£1) UBS Securities Australia Ltd Sell 1,222 (£2) UBS Securities Australia Ltd Sell 2,222 (£2) UBS Securities Australia Ltd Sell 3,324 (£2) UBS Securities Australia Ltd Sell 7,773 (£2) UBS Securities Australia Ltd Sell 7,773 (£2) UBS Securities Australia Ltd Sell 2,265,500 (£2),000 UBS Securities Australia Ltd Sell 2,265,500 (£20,000) UBS Securities Australia Ltd Sell 3,654 (£3) UBS Securities Australia Ltd Sell 3,654 (£403) UBS Securities Australia Ltd Sell 3,654 (£404) UBS Securities Australia Ltd Sell 3,654 (£404) UBS Securities Australia Ltd Sell 3,654 (£404) UBS Securities Australia Ltd Sell 3,664 (£404) UBS Securities Australia Ltd Sell 3,675 (£404) UBS Securities	12-May-14	UBS Securities Australia Ltd	Sell	363	(40)	Ordinary
UBS Securities Australia Ltd Sell 190 (17) UBS Securities Australia Ltd Sell 2,222 (245) UBS Securities Australia Ltd Sell 2,222 (125) UBS Securities Australia Ltd Sell 1,373 (1,028) UBS Securities Australia Ltd Sell 2,262,500 (25,000) UBS Securities Australia Ltd Sell 1,773 (1,028) UBS Securities Australia Ltd Sell 2,262,500 (25,000) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 3,075 (1,000) UBS Securities Australia Ltd Sell 3,075 (1,000) UBS Securities Australia Ltd Sell 3,075 (1,000) UBS Securities Australia Ltd Sell 3,070 (1,000) UBS Securities Australia Ltd Sell 3,070 (1,000) UBS	12-May-14	ODS SECULIES AUSTRAIG LIG	Sell Sell	81/8	(410)	Ordinary
UBS Securities Australia Ltd Sell 2,222 (2.5) UBS Securities Australia Ltd Sell 2,324 (1,031) UBS Securities Australia Ltd Sell 7,773 (857) UBS Securities Australia Ltd Sell 7,773 (873) UBS Securities Australia Ltd Sell 7,272 (1,531) UBS Securities Australia Ltd Sell 7,272 (1,531) UBS Securities Australia Ltd Sell 2,225,500 (1,500) UBS Securities Australia Ltd Sell 3,525 (1,500) UBS Securities Australia Ltd Sell 3,655 (1,000) UBS Securities Australia Ltd Sell 3,675 (1,000)	12-May-14	UBS Securities Australia Ltd	Je.	190	(21)	Ordinary
UBS Securities Australia Ltd Sell 2,222 (1,028) UBS Securities Australia Ltd Sell 7,773 (1,028) UBS Securities Australia Ltd Sell 7,773 (1,631) UBS Securities Australia Ltd Sell 1,4723 (1,631) UBS Securities Australia Ltd Sell 1,4723 (1,631) UBS Securities Australia Ltd Sell 1,4723 (1,631) UBS Securities Australia Ltd Sell 2,262,500 (250,000) UBS Securities Australia Ltd Sell 3,000 (393) UBS Securities Australia Ltd Sell 3,000 (393) UBS Securities Australia Ltd Sell 3,000 (393) UBS Securities Australia Ltd Sell 3,000 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 1,773 (1,000)	12-May-14	UBS Securities Australia Ltd	Sell	6	(1)	Ordinary
UBS Securities Australia Ltd Sell 9.324 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	12-May-14	UBS Securities Australia Ltd	Sell	2,222	(245)	Ordinary
UBS Securities Australia Ltd Sell 7,773 (1,028) UBS Securities Australia Ltd Sell 7,773 (872) UBS Securities Australia Ltd Sell 1,4793 (1,631) UBS Securities Australia Ltd Sell 2,262,500 (250,000) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 5,675 (1,500) UBS Securities Australia Ltd Sell 5,778 (1,500) UBS Secur	12-May-14	UBS Securities Australia Ltd	Seil	6	î)	Ordinary
UBS Securities Australia Ltd Sell 1,773 (1871) UBS Securities Australia Ltd Sell 2,262,500 (250,000) UBS Securities Australia Ltd Sell 2,262,500 (250,000) UBS Securities Australia Ltd Sell 3,555 (403) UBS Securities Australia Ltd Sell 888 (110) UBS Securities Australia Ltd Sell 888 (100) UBS Securities Australia Ltd Sell 3,653 (404) UBS Securities Australia Ltd Sell 3,654 (404) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 3,656 (100) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,133 (1,200) UBS Securities Australia Ltd Sell 1,278 (1,200) UBS Securities Australia Ltd Sell 1,364 (1,200) UBS S	12-May-14	UBS Securities Australia Ltd	Sell	9,324	(1,028)	Ordinary
UBS Securities Australia Ltd Sell 2,262,500 (1,631) UBS Securities Australia Ltd Sell 2,262,500 (250,000) UBS Securities Australia Ltd Sell 1,415 (1,56) UBS Securities Australia Ltd Sell 1,415 (1,52) UBS Securities Australia Ltd Sell 3,020 (333) UBS Securities Australia Ltd Sell 3,020 (403) UBS Securities Australia Ltd Sell 3,020 (1,000) UBS Securities Australia Ltd Sell 3,020 (1,000) UBS Securities Australia Ltd Sell 3,020 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,261 (1,000) UBS Securities Australia Ltd Sell 1,260 (1,000)	12-May-14	UBS Securities Australia Ltd	Seli	2,773	(857)	Ordinary
UBS Securities Australia Ltd Sell 2.262,500 (250,000) UBS Securities Australia Ltd Sell 1,415 (150) UBS Securities Australia Ltd Sell 1,415 (150) UBS Securities Australia Ltd Sell 898 (10) UBS Securities Australia Ltd Sell 898 (10) UBS Securities Australia Ltd Sell 3,652 (403) UBS Securities Australia Ltd Sell 3,652 (403) UBS Securities Australia Ltd Sell 3,652 (403) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,550) UBS Securities Australia Ltd Sell 1,778 (1,550) UBS Securities Australia Ltd Sell 1,778 (1,550) UBS Securities Australia Ltd Sell 5,718 (1,550) UBS Securities	12-May-14	UBS Securities Australia Ltd	Sell	14,793	(1,631)	Ordinary
UBS Securities Australia Ltd Sell 2.262,500 (750) UBS Securities Australia Ltd Sell 472 (152) UBS Securities Australia Ltd Sell 3655 (403) UBS Securities Australia Ltd Sell 898 (199) UBS Securities Australia Ltd Sell 3654 (110) UBS Securities Australia Ltd Sell 3652 (403) UBS Securities Australia Ltd Sell 3656 (403) UBS Securities Australia Ltd Sell 9070 (1000) UBS Securities Australia Ltd Sell 1,338 (1,250) UBS Securities Australia Ltd Sell 1,659 (1,000) UBS Securities Australia Ltd Sell 1,650 (1,000) UBS Securities Australia L	12-May-14	UBS Securities Australia Ltd	Sell	2,262,500	(250,000)	Ordinary
UBS Securities Australia Ltd Sell 415 (156) UBS Securities Australia Ltd Sell 472 (432) UBS Securities Australia Ltd Sell 958 (403) UBS Securities Australia Ltd Sell 956 (110) UBS Securities Australia Ltd Sell 3020 (333) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,267 (1,000) UBS Securities Australia Ltd Sell 1,267 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 5,771 (1,500) UBS Securities Au	12-May-14	UBS Securities Australia Ltd	Sell	2,262,500	(250,000)	Ordinary
UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 898 (110) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 9,075 (1,211) UBS Securities Australia Ltd Sell 9,075 (1,200) UBS Securities Australia Ltd Sell 9,070 (1,339) UBS Securities Australia Ltd Sell 9,070 (1,300) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securi	12-May-14	UBS Securities Australia Ltd	Sell	1,415	(156)	Ordinary
UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 998 (100) UBS Securities Australia Ltd Sell 3,654 (404) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 1,651 (1,000) UBS Securities Australia Ltd Sell 1,651 (1,000) UBS Securities Australia Ltd Sell 1,651 (1,000) UBS Securi	12-May-14	UBS Securities Australia Ltd	Sell	472	(52)	Ordinary
UBS Securities Australia Ltd Sell 898 (199) UBS Securities Australia Ltd Sell 996 (110) UBS Securities Australia Ltd Sell 3,020 (333) UBS Securities Australia Ltd Sell 3,020 (333) UBS Securities Australia Ltd Sell 3,055 (400) UBS Securities Australia Ltd Sell 0,075 (1,000) UBS Securities Australia Ltd Sell 0,070 (1,000) UBS Securities Australia Ltd Sell 0,070 (1,000) UBS Securities Australia Ltd Sell 0,070 (1,000) UBS Securities Australia Ltd Sell 1,1,738 (1,250) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securi	12-May-14	UBS Securities Australia Ltd	Sell	3,655	(403)	Ordinary
UBS Securities Australia Ltd Sell 958 (110) UBS Securities Australia Ltd Sell 3,654 (404) UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,1,338 (1,250) UBS Securities Australia Ltd Sell 1,78 (1,500) UBS Securities Australia Ltd Sell 1,78 (1,500) UBS Securities Australia Ltd Sell 1,78 (1,500) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Sec	12-May-14	UBS Securities Australia Ltd	lleS.	868	(66)	Ordinary
UBS Securities Australia Ltd Sell 3,654 (404) UBS Securities Australia Ltd Sell 3,020 (333) UBS Securities Australia Ltd Sell 3,655 (4,03) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,788 (1,250) UBS Securities Australia Ltd Sell 1,778 (1,600) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,000 (1,000) UBS	12-May-14	UBS Securities Australia Ltd	Sell	866	(110)	Ordinary
UBS Securities Australia Ltd Sell 3,020 (333) UBS Securities Australia Ltd Sell 3,655 (400) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,261 (1,39) UBS Securities Australia Ltd Sell 1,778 (1,50) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Sec	12-May-14	UBS Securities Australia Ltd	Sell	3,664	(404)	Ordinary
UBS Securities Australia Ltd Sell 3,655 (403) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 10,984 (1,211) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,78 (1,550) UBS Securities Australia Ltd Sell 1,78 (1,500) UBS Securities Australia Ltd Sell 1,78 (1,550) UBS Securities Australia Ltd Sell 1,78 (1,500) UBS Securities Australia Ltd Sell 1,78 (1,500) UBS Securities Australia Ltd Sell 1,691 (2,171) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,614 (2,000) UBS Securities Australia Ltd Sell 1,621 (1,000) UBS	12-May-14	UBS Securities Australia Ltd	Sell	3,020	(333)	Ordinary
UBS Securities Australia Ltd Sell 9,156 (900) UBS Securities Australia Ltd Sell 9,075 (1,000) UBS Securities Australia Ltd Sell 6,676 (736) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 38,611 (4,527) UBS Securities Australia Ltd Sell 1,261 (1,500) UBS Securities Australia Ltd Sell 1,778 (1,550) UBS Securities Australia Ltd Sell 1,778 (1,550) UBS Securities Australia Ltd Sell 1,652 (62) UBS Securities Australia Ltd Sell 1,667 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,000 (1,000) UBS	12-May-14	UBS Securities Australia Ltd	lles	3,655	(403)	Ordinary
UBS Securities Australia Ltd Sell 10,984 (1,000) UBS Securities Australia Ltd Sell 10,984 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,261 (1,39) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 2,168 (2,21) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) UB	12-May-14	UBS Securities Australia Ltd	Sell	8,168	(006)	Ordinary
UBS Securities Australia Ltd Sell 10,984 (1,211) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,261 (1,39) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,78 (1,200) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,078 (1,000) UBS Securities Australia Ltd Sell 1,071 (1,000) <td< td=""><td>12-May-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>9.075</td><td>(1,000)</td><td>Ordinary</td></td<>	12-May-14	UBS Securities Australia Ltd	Sell	9.075	(1,000)	Ordinary
UBS Securities Australia Ltd Sell 6,676 (736) UBS Securities Australia Ltd Sell 39,70 (1,000) UBS Securities Australia Ltd Sell 30,70 (1,000) UBS Securities Australia Ltd Sell 1,261 (1,39) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,5691 (2,171) UBS Securities Australia Ltd Sell 2,166 (2,39) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 5,000 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,000) U	12-May-14	UBS Securities Australia Ltd	Seil	10.984	(1.211)	Ordinary
UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,267 (1,000) UBS Securities Australia Ltd Sell 1,267 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,657 (1,000) UBS Securities Australia Ltd Sell 2,168 (2,171) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,96) UB	12-Mav-14	UBS Securities Australia Ltd	Jay.	6676	(984)	Ordinary
UBS Securities Australia Ltd Sell 38,6171 (1,520) UBS Securities Australia Ltd Sell 1,000 (1,000) UBS Securities Australia Ltd Sell 1,138 (1,250) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 1,6591 (2,177) UBS Securities Australia Ltd Sell 1,6691 (2,177) UBS Securities Australia Ltd Sell 1,814 (2,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,000 (1,000)	12-May-14	I IRS Securities Australia 1td	3	0,00,0	(000)	Ordinary
UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,261 (1,300) UBS Securities Australia Ltd Sell 1,778 (1,250) UBS Securities Australia Ltd Sell 1,778 (1,500) UBS Securities Australia Ltd Sell 1,9,691 (2,171) UBS Securities Australia Ltd Sell 2,168 (3,29) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 5,623 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,96) U	12-May-14	I IBC Con writing Australia 144	la Coll	2000	(2000)	Ordinary
UBS Securities Australia Ltd Sell 1,1261 (1,350) UBS Securities Australia Ltd Sell 1,1,78 (1,550) UBS Securities Australia Ltd Sell 1,778 (1,550) UBS Securities Australia Ltd Sell 1,778 (1,50) UBS Securities Australia Ltd Sell 1,614 (2,171) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,621 (1,78) UBS Securities Australia Ltd Sell 1,621 (1,78) UBS Securities Australia Ltd Sell 1,738 (1,60) UBS Securities Australia Ltd Sell 1,738 (1,60) U	12 14-14	Obs securities Abstratia Ltd	E C	10,00	(4,257)	Orginary
UBS Securities Australia Ltd Sell 1,261 (139) UBS Securities Australia Ltd Sell 1,778 (136) UBS Securities Australia Ltd Sell 1,778 (136) UBS Securities Australia Ltd Sell 1,671 (2,171) UBS Securities Australia Ltd Sell 2,168 (2,171) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 5,623 (620) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 1,778 (1,00) UBS Securit	12-ivlay-14	UBS Securities Australia Ltd	Sell	0/0'6	(1,000)	Ordinary
UBS Securities Australia Ltd Sell 1,338 (1,250) UBS Securities Australia Ltd Sell 1,778 (196) UBS Securities Australia Ltd Sell 1,651 (2,171) UBS Securities Australia Ltd Sell 2,168 (2,171) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 960 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 5,020 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,99) UBS Securities Australia Ltd Sell 1,778 (1,00) UBS Securities Australia Ltd Sell 915 (1,00) UBS Securi	12-May-14	UBS Securities Australia Ltd	Sell	1,261	(139)	Ordinary
UBS Securities Australia Ltd Sell 1,778 (156) UBS Securities Australia Ltd Sell 552 (62) UBS Securities Australia Ltd Sell 19,691 (2,177) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,38) UBS Securities Australia Ltd Sell 1,6217 (1,38) UBS Securities Australia Ltd Sell 1,6217 (1,58) UBS Securities Australia Ltd Sell 1,6217 (1,58) UBS Securities Australia Ltd Sell 1,678 (1,60) UBS Securities Australia Ltd Sell 1,678 (1,60) UBS Securities Australia Ltd Sell 1,600 (1,60) UBS Securities Australia Ltd Sell 1,600 (1,60) UBS Secur	12-May-14	UBS Securities Australia Ltd	Sell	11,338	(1,250)	Ordinary
UBS Securities Australia Ltd Sell 19,691 (62) UBS Securities Australia Ltd Sell 19,691 (2,171) UBS Securities Australia Ltd Sell 2,168 (1,200) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 5,023 (6,20) UBS Securities Australia Ltd Sell 5,623 (6,20) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 9,15 (1,00) UBS Securities Australia Ltd Sell 9,15 (1,00) UBS Securities Australia Ltd Sell 9,15 (1,00) UBS Securit	12-May-14	UBS Securities Australia Ltd	Sell	1,778	(196)	Ordinary
UBS Securities Australia Ltd Sell 19,691 (2,171) UBS Securities Australia Ltd Sell 2,168 (2,39) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 2,09 (2,3) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 915 (1,00) UBS Securities Australia Ltd Sell 915 (1,00) UBS Securities Australia Ltd Sell 915 (1,00) UBS Securities Australia Ltd Sell 1,501 (1,00) UBS Securities A	12-May-14	UBS Securities Australia Ltd	Sell	295	(9)	Ordinary
UBS Securities Australia Ltd Sell 2,168 (739) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,778 (1,788) UBS Securities Australia Ltd Sell 1,600 (200,000) UBS Securities Australia Ltd Sell 1,800,000 (200,000) UBS Securities Australia Ltd Sell 1,800,000 (200,000) UBS Securities Australia Ltd Sell 9,367 (100) UBS Securities Australia Ltd Sell 9,15 (100) UBS Securities Australia Ltd Sell 9,11 (100) UBS	12-May-14	UBS Securities Australia Ltd	Sell	169'61	(2,171)	Ordinary
UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 1,814 (200) UBS Securities Australia Ltd Sell 980 (1,000) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 5,623 (23) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 1,808,000 (200,000) UBS Securities Australia Ltd Sell 3,367 (100) UBS Securities Australia Ltd Sell 951 (100) UBS Securities Australia Ltd Sell 951 (100) UBS Securities Australia Ltd Sell 1,501 (160) UBS Securities Australia Ltd Sell 1,501 (160) UBS Securities Australia Ltd Sell 1,501 (160)	12-May-14	UBS Securities Australia Ltd	Sell	2,168	(239)	Ordinary
UBS Securities Australia Ltd Sell 1,814 (200) UBS Securities Australia Ltd Sell 980 (108) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 5,623 (23) UBS Securities Australia Ltd Sell 1,778 (1,788) UBS Securities Australia Ltd Sell 1,778 (1,466) UBS Securities Australia Ltd Sell 1,778 (1,66) UBS Securities Australia Ltd Sell 1,878 (100) UBS Securities Australia Ltd Sell 91,5 (100) UBS Securities Australia Ltd Sell 91,5 (100) UBS Securities Australia Ltd Sell 91,5 (100) UBS Securities Australia Ltd Sell 1,501 (160)	12-May-14	UBS Securities Australia Ltd	Sell	9.070	(1,000)	Ordinary
UBS Securities Australia Ltd Sell 960 (108) UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 2,09 (7,31) UBS Securities Australia Ltd Sell 1,6,217 (1,788) UBS Securities Australia Ltd Sell 1,778 (1,989) UBS Securities Australia Ltd Sell 1,808,000 (200,000) UBS Securities Australia Ltd Sell 3,367 (160) UBS Securities Australia Ltd Sell 915 (100) UBS Securities Australia Ltd Sell 915 (100) UBS Securities Australia Ltd Sell 951 (100) UBS Securities Australia Ltd Sell 951 (100) UBS Securities Australia Ltd Sell 1,501 (160) UBS Securities Australia Ltd Sell 1,501 (160)	12-May-14	UBS Securities Australia Ltd	Sell	1.814	(200)	Ordinary
UBS Securities Australia Ltd Sell 9,070 (1,000) UBS Securities Australia Ltd Sell 209 (23) UBS Securities Australia Ltd Sell 5,623 (620) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 1,778 (1,96) UBS Securities Australia Ltd Sell 1,808,000 (200,000) UBS Securities Australia Ltd Sell 3,367 (100) UBS Securities Australia Ltd Sell 951 (100) UBS Securities Australia Ltd Sell 1,501 (165) UBS Securities Australia Ltd Sell 1,501 (164) UBS Securities Australia Ltd Sell 1,501 (164) UBS Securities Australia Ltd Sell 1,501 (164)	12-Mav-14	UBS Securities Australia Ltd	Seil	086	(108)	Ordinary
UBS Securities Australia Ltd Sell 200 (7.33) UBS Securities Australia Ltd Sell 5.623 (6.20) UBS Securities Australia Ltd Sell 1.5,217 (1.788) UBS Securities Australia Ltd Sell 1,778 (1.96) UBS Securities Australia Ltd Sell 1,878 (1.96) UBS Securities Australia Ltd Sell 3,367 (100) UBS Securities Australia Ltd Sell 91.5 (100) UBS Securities Australia Ltd Sell 91.5 (100) UBS Securities Australia Ltd Sell 1,501 (160) Securities Australia Ltd Sell (160) (160)	12-May-14	UBS Securities Australia 11d	Sell	020 6	(000)	Ordinary
UBS Securities Australia Ltd Sell 5,623 (620) UBS Securities Australia Ltd Sell 1,788 (1,788) UBS Securities Australia Ltd Sell 1,808,000 (200,000) UBS Securities Australia Ltd Sell 3,367 (368) UBS Securities Australia Ltd Sell 915 (100) UBS Securities Australia Ltd Sell 961 (100) UBS Securities Australia Ltd Sell 1,501 (165) UBS Securities Australia Ltd Sell (37)	12-Mav-14	UBS Securities Australia Ltd	lleS	906	(53)	Ordinary
UBS Securities Australia Ltd Sell 1,778 (1,788) UBS Securities Australia Ltd Sell 1,778 (196) UBS Securities Australia Ltd Sell 1,808,000 (200,000) UBS Securities Australia Ltd Sell 3,367 (100) UBS Securities Australia Ltd Sell 951 (100) UBS Securities Australia Ltd Sell 951 (105) UBS Securities Australia Ltd Sell 1,501 (164) UBS Securities Australia Ltd Sell 888 (97)	12-May-14	UBS Securities Australia I1d	Sell	5,623	(620)	Ordinan
UBS Securities Australia Ltd Sell 1,778 (159) UBS Securities Australia Ltd Sell 1,808,000 (200,000) UBS Securities Australia Ltd Sell 3,367 (369) UBS Securities Australia Ltd Sell 915 (100) UBS Securities Australia Ltd Sell 961 (105) UBS Securities Australia Ltd Sell 1,501 (164) UBS Securities Australia Ltd Sell 888 (37)	12-Mav-14	UBS Securities Australia Ltd	les	16.217	(1 788)	Ordinary
UBS Securities Australia Ltd Sell 1,808,000 (200,000) UBS Securities Australia Ltd Sell 3,367 (368) UBS Securities Australia Ltd Sell 915 (100) UBS Securities Australia Ltd Sell 961 (105) UBS Securities Australia Ltd Sell 1,501 (164) UBS Securities Australia Ltd Sell (97) (164)	12-May-14	I IBS Securities Australia I to	Sell	1 778	(106)	Ordinan
UBS Securities Australia Ltd Sell 1,500,000 UBS Securities Australia Ltd Sell 915 (100) UBS Securities Australia Ltd Sell 961 (105) UBS Securities Australia Ltd Sell 1,501 (164) UBS Securities Australia Ltd Sell 1,501 (164)	12-May-14	LIBS Securities Australia Ltd	Sell	1 808 000	(000 000)	Ordings
UBS Securities Australia Ltd Sell Sell (100) UBS Securities Australia Ltd Sell Sell (100) UBS Securities Australia Ltd Sell (101) UBS Securities Australia Ltd Sell (104) UBS Securities Australia Ltd Sell (100) UBS Securities Australia Ltd Sell (100) UBS Securities Australia Ltd Sell (100) UBS Securities Australia Ltd (100) UBS Securities Australia L	12-May-14	BC Counties Australia 144	Coll	736 6	(000,002)	Ordinary
Obs. Securities Australia Ltd Sell 913 (100) UBS Securities Australia Ltd Sell 961 (105) UBS Securities Australia Ltd Sell 961 (164) UBS Securities Australia Ltd Sell 15.01 (164) UBS Securities Australia Ltd Sell 888 (97)	12 May 14	Upo Consider August and Ltd	nec o	796,6	(300)	Orginary
UBS Securities Australia Ltd Sell 951 (105) UBS Securities Australia Ltd Sell 1,501 (164) UBS Securities Australia Ltd Sell 888 (97)	12-May-14	UBS Securities Australia Lid	Sell	315	(001)	Ordinary
UBS Securities Australia Ltd Sell 1,501 (164) UBS Securities Australia Ltd Sell 888 (97)	12-May-14	UBS Securities Australia Ltd	Sell	961	(105)	Ordinary
UBS Securities Australia Ltd Sell Sell (97)	12-May-14	UBS Securities Australia Ltd	Sell	1,501	(164)	Ordinary
	2-May-14	UBS Securities Australia Ltd	Sell	888	(26)	Ordinary

5-May-14	I IBS Socientiae Australia Ltd	a c	.00	100	,
5-May-14	UBS Securities Australia Itd	Riiv	102,1	132	Ordinary
-May-14	UBS Securities Australia Ltd	Buv	46	1 11	Ordinary
15-May-14	ILIBS Securities Australia Ltd	Bank	27.0	200	O Gilling
5-May-14	I IRC Constition Australia 14d	100	17	1	Grainary
5-May-14	I IRS Securities Australia 144	ind G	190	30	Orginary
5-May-14	I IRS Securities Australia (4d	in a	970	7	Orginary
5-May-14	I IRS Securities Australia I to	in a	000000	7 0000	Cranaly
5-May-14	I IRS Securities Australia I to	2 8	16 000	10,01	Collinary
5-May-14	UBS Securities Australia Ltd	8.2	11 284	1,00/	Ordinary
5-May-14	I IRS Securities Australia I to	Birch	13 650	2	ordinary.
5-May-14	UBS Securities Australia Ltd	Buck	00000	246	Orderially
May-14	UBS Securities Australia 11d	Briv	7 863	040	Ciding
15-May-14	UBS Securities Australia Ltd	Birk	2 140	100	Ordinary
May 14	Life Consister Americal Life	909	3,149	240	Organiary
E-4424-14	I IDC Cocurition Australia 344	Pag.	/7		Ordinary
May 14	Upo Cocupios Australia Ltd	Buy	ה פיני		Ordinary
D-INIdy-14	Obsideration of the control of the c	Buy	66,830	7,344	Ordinary
4 Velvi-C	OBS Securities Australia Lto	Buy	57,671	6,341	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	5,824	640	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	6	-	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	27	3	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	3,376	371	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	1,274	140	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	36	4	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	11,284	1,240	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	72,645	7,983	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	1,238	136	Ordinary
5-May-14	UBS Securities Australia Ltd	Bay	36	7	Ordinary
5-May-14	UBS Securities Australia Ltd	Bity	2,697	705	Ordinary
5-Mav-14	UBS Securities Australia Ltd	Briv	5,642	620	Coning
5-May-14	I IRS Securities Australia I to	Biss	0		in in
May 14	Del Cicator Australia	600	6	- 4	Ciurally F
5-May-14	I IBC Contribute Australia 14d	Special Company	1,5/4	1/3	Orginary
Man 14	Clos occultures Australia LIO	Buy	87/	08	Ordinary
May-14	UBS Securities Australia Ltd	Buy	774	82	Ordinary
D-INIdy-14	UBS Securities Australia Ltd	Buy	80,535	8,850	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	4,714	518	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	66,849	7,346	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	29,903	3,286	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	17,199	1,890	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	6	-	Ordinary
5-May-14	UBS Securities Australia Ltd	Buv	7.862	864	Ordinary
5-Mav-14	UBS Securities Australia 11d	Birk	3 140	346	Ording
5. May 14	I DC Counties Australia 144	600	5,149	040	Oldinary
14 ay 17	ODS Securities Australia Ltd	, and	3,149	345	Ordinary
D-Ividy-14	Obsideration Australia Ltd	Buy	3,149	346	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	4,477	492	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	4,323	475	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	19,938	2,191	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	46	2	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	32,778	3,602	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	1,065	117	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	16.325	1.794	Ordinary
-May-14	UBS Securities Australia Ltd	Buv	2.029	223	Ordinary
5-May-14	UBS Securities Australia Ltd	Buv	255	28	Ordinary
5-Mav-14	LIBS Securities Australia Ltd	Buy	8 964	200	Ordinan
5-May-14	I IBS Societies Australia I to	9.50	50,500	200	Cidillary
5-May-14	I IRS Securities Australia Ltd	Bin	0.100	1,000	Oldling
C Learn 14	The Committee Australia Ltd	, and	001,2	000,1	Ordinary
D-IVIdy-14	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
- Kelai	Obs securities Australia Liu	euy	1/3	5	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	11,202	1,231	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	29,930	3,289	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
- 1	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
	UBS Securities Australia Ltd	Buy	118	43	, and a
				2	Cramary
15-May-14	UBS Securities Australia Ltd	Buv	3.149	346	Ordinary

UBS Securities Australia Ltd Sell 19. UBS Securities Australia Ltd Sell 41. UBS Securities Australia Ltd Sell 41. UBS Securities Australia Ltd Sell 19. UBS Securities Australia Ltd Buy 262. UBS Securities Australia Ltd Buy 262. UBS Securities Australia Ltd Sell 51. UBS Securities Australia Ltd Buy 262. UBS Securities Australia Ltd Buy 27. UBS Securities Australia Ltd Buy 27. UBS Securities Australia Ltd Buy 37.	 	urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd London Branch London Branch London Branch London Branch London Branch urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Sel	13,953 1,001 1,010	(1,526) (4,482) (564) (13,106) (1,584) (1,584) (28,56) (24,566) (3,500) (19,246) (19,246) (19,246) (100,524) (100,52	
1985 Securities Autoratival Ltd 5441 119,020 14,101 14,1422 14,1	 	unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd London Branch London Br	Sell Sell Buy Sell Buy Sell Buy Sell Buy Sell Buy Buy Buy Buy Sell Buy Buy Buy Sell Sell Buy Buy Buy Sell Sell Sell Sell Sell Sell Sell Sel	1,001 1,100 1,100 1,100 1,000	(4.47) (1.564) (1.564) (1.586) (1.586) (28.56) (28.56) (3.456) (5.60) (5.60) (5.60) (5.60) (5.60) (5.60) (6.166) (6.16	
Bis Securities Autorials Ltd Self 11,12,00 11,1	 	unities Australia Ltd urities Australia Ltd	Seli	5,161 119,920 12,482 224,739 430,219 696,889 3,926 4,337 11,73 211 211 21,409 11,609 1	(13.06) (13.06) (13.06) (13.06) (13.06) (13.06) (13.06) (13.06) (13.06) (10.05) (10.05) (10.05) (10.05) (10.05) (10.05) (10.05)	
Bis Securities Autoralia Ltd Buy 19,200 (13,100)	 	urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd London Banch London Banch London Banch London Banch urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Self Se	119,920 14,482 327,489 40,219 17,598 86,869 17,72 47,372 211 47,372 47,372 40,931 183,994 110,493 410,493 110,493 247,981 281,100 91,400 91,400 247,981 281,400	(13,105) (1,584) (1,584) (1,584) (1,584) (24,5019) (1,924	
1975 Securities Autoralia Ltd 1979 1973 1975 1975 1975 Securities Autoralia Ltd 1974 1975	 	urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd London Branch London Branch London Branch London Branch urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Self Buy	14,482 224,789 224,789 224,789 224,789 175,985 177,174 271 271 271 43,280 19,483 19,48	(1,584) 35,867 24,567 (47,019) (19,246) (19,246) (19,246) (19,246) (19,246) (10,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524)	
Bis Securities Australia Ltd Buy 22,580 2,580 LBS Securities Australia Ltd Sell 22,475 22,450 (28,551) LBS Securities Australia Ltd Sell 175,985 (10,246) LBS Securities Australia Ltd Buy 22,475 (10,246) LBS Securities Australia Ltd Buy 3,225 (10,226) LBS Securities Australia Ltd Buy 3,225 (10,226) LBS Securities Australia Ltd Buy 3,225 (10,226) LBS Securities Australia Ltd Buy 3,225 (10,227) LBS Securities Australia Ltd Buy 2,236 (20,227)	 	urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd London Branch London Branch London Branch London Branch urities Australia Ltd	8 8 9 8 9 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9	262,430 224,735 430,219 51,487 115,985 696,895 696,895 3,926 47,372 211 43,73 40,921 183,944 149,344 19,948 19,483 1100 91,100 91,400 11,000 91,400 9	35,861 (47,019) (5,600) (5,600) (5,600) (5,600) (5,160) (5,160) (5,160) (1,00,524) (1,00	
Bis Securities Australia Ltd Self 224,250 (285,551) Dis Securities Australia Ltd Self 24,719 (47,019) Dis Securiti	 	urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd London Branch Lifes Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Sell Buy Sell Sell Sell Buy Buy Buy Buy Sell Sell Sell Sell Sell Sell Sell Suy	222,430 224,739 430,219 51,487 51,885 656,889 3,926 47,372 211 40,271 183,994 110,493 410,493 410,493 110,	(10,5,50) (10,5,60) (10,5,60) (10,5,60) (10,5,40) (10,5,	
URS Securities Australia Ltd Bly 224,759 24,556 UBS Securities Australia Ltd Sell 175,988 (47,1019) UBS Securities Australia Ltd Sell 175,988 (19,246) UBS Securities Australia Ltd Sell 175,989 (17,1019) UBS Securities Australia Ltd Bly 5,266 7,246 UBS Securities Australia Ltd Bly 6,676 7,271 UBS Securities Australia Ltd Bly 6,671 7,271 UBS Securities Australia Ltd Bly 6,672 7,273 UBS Securities Australia Ltd Bly 6,17,174 (10,522) UBS Securities Australia Ltd Bly 7,174 (10,522)	 	untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd London Branch London Branch London Branch London Branch London Branch Lintes Australia Ltd untites Australia Ltd	Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	224,759 430,219 5,1487 175,988 696,889 1326 47,377 47,377 47,377 114,373 118,394 110,483 110,483 110,483 110,483 110,483 110,483 110,483 110,483 110,483 110,483 110,483 110,483 110,483 110,483 114,573 114,5	(4,701) (19,245) (19,245) (19,245) (19,245) (19,245) (10,242) (10,524) (10,524) (10,524) (10,524) (10,524) (10,524) (10,524) (10,524)	
URS Securities Australia Ltd Self 450,219 (47,019) URS Securities Australia Ltd Self 15,487 (5,000) URS Securities Australia Ltd Self 17,548 (15,246) URS Securities Australia Ltd Buy 5,000 (4,019) URS Securities Australia Ltd Self 7,174 (10,022) URS Securities Australia Ltd Buy 3,17 (10,022) URS Securities Australia Ltd Buy 1,17 (10,022) URS Securities Australia Ltd Buy 1,10 (10,000) URS Securities Australia Ltd Buy 1,10 (10,000) URS Securities Australia Ltd Buy 1,10 (10,000) URS Securities Australia Ltd Buy 2,10 (2,100) URS Securities Australia Ltd Buy 2,10 (10,000) URS	 	unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd London Branch London Br	Self Self Self Self Buy Buy Self	400.219 51.487 17.586 696.868 3.926 47.372 21.174 917,174	(4/7)(19) (5,000) (5,166) (5,166) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524) (100,524)	
UBS Securities Australia Ltd Sull 17,587 (15,500) BSS Securities Australia Ltd Buy 6,566,869 76,168 BSS Securities Australia Ltd Buy 5,305 7,174 BSS Securities Australia Ltd Buy 2,31 2,3 BS Securities Australia Ltd Buy 2,31 4,7 BSS Act London Branch Buy 2,31 4,7 BSS Act London Branch Buy 9,17,174 (10,5224) BSS Act London Branch Buy 9,17,174 (10,527) BSS Act London Branch Buy 9,17,174 (10,527) BSS Securities Australia Ltd Buy 9,17,174 (10,527) BSS Securities Australia Ltd Buy 9,17,104 (10,527) BSS Securities Australia Ltd Buy 1,100 (10,500) BSS Securities Australia Ltd Buy 2,100 (10,500) BSS Securities Australia Ltd Buy 2,100 (10,500) BSS Securities Australia Ltd Buy 2,100 (10,500) BSS Securities Australia Ltd	 - - - - - - - - - - -	urities Australia Ltd urities Australia Ltd urities Australia Ltd bal Asset Management rities Australia Ltd London Branch London	Sell Sell Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	51,487 696,886 696,886 7,326 47,372 211 211 21,483 18,394 18,394 18,394 19,483 19,483 19,483 10,493 21,100 91,100 91,403 10,493 10,493 10,493 10,493 10,493 10,493 11,004 11,	(19246) (19246) (19246) (5,168) (23 (23 (100,524) (100,524) (100,524) (4,483) (4,483) (100,524) (100,524) (100,524) (100,524)	
Bis Securities Australia Ltd Sell 175,285 (15,246) 1085 Securities Australia Ltd Buy 5,985 76,168 1085 Gobal Assert Indiagement Buy 5,372 5,166 1085 Gobal Assertifies Ltd Swil 4,377 1,377 1,57 1085 Gobal Assertifies Ltd Buy 4,377 1,37 1,57 1085 Securities Australia Ltd Buy 4,37 1,00 1,00 1085 Securities Australia Ltd Buy 9,11,10 1,00 2,00 1085 Securities Australia Ltd Buy 9,11,00 1,00 1,00 1085 Securities Australia Ltd Buy 9,11,00 1,00 1,00 1085 Securities Australia Ltd Buy 9,14,71 1,15 1,15 1085 Securities Australia Ltd Buy 9,14 1,15 1,15 1085 Securities Australia Ltd Buy 2,10 1,10 1,10 1085 Securities Australia Ltd Buy 2,10 1,10 1,10 1085 Securities Australia Ltd Buy 2,10	 	unities Australia Ltd Turties Australia Ltd Jea Asset Management Turties Australia Ltd London Branch Londo	Self ii Buy Buy Self ii Buy Self ii Self ii Self ii Self ii Suy	175,988 96,889 3,326 47,375 41,375 41,317 917,114 919,483 183,994 110,493 410,494 91,400 91,400 91,400 91,400 91,400 91,400 91,400 91,400 91,400 91,400 91,400	(19,246) (19,246) 76,168 (2,166) 23 (100,524) (100,524) (4,483) (2,060) (16,377)	
UBS Securities Australia Ltd Bly G96,699 76,168 UBS Scarrifies Australia Ltd Bly 3,726 4,78 UBS Scarrifies Australia Ltd Bly 3,726 4,78 UBS Scarrifies Australia Ltd Bly 3,717 4 UBS Scarrifies Australia Ltd Bly 3,717 4 UBS Scarrifies Australia Ltd Bly 9,17,174 4 UBS Scarrifies Australia Ltd Bly 9,17,174 1,000 UBS Scarrifies Australia Ltd Bly 9,17,174 1,000 UBS Scarrifies Australia Ltd Bly 9,17,617 1,000 UBS Scarrifies Australia Ltd Bly 1,000 1,000 UBS Scarrifies Australia Ltd Bly 2,506 2,71 UBS Scarrifies Australia Ltd Bly 2,607 1,000 UBS Scarrifies Australia Ltd Bly 2,607 1,000 UBS Scarrifies Australia Ltd Bly 2,607 2,607 UBS Scarrifies Australia Ltd Bly 2,607 2,607 UBS Scarrifies Australia Ltd <	 	unties Australia Ltd Asser Management urities Australia Ltd Asser Management urities Australia Ltd London Branch L	Buy Buy Buy Buy Buy Buy Sell Sell Sell Sell Buy Buy Buy Buy Buy	996,896 3,926 4,372 211 211 431 917,174 919,483 183,994 183,994 194,348 91,100 91,100 91,100 91,100 91,100 91,100 91,100 91,100 91,100 91,100	(5,166) (2,166) (3,166) (4,166) (100,524) (1,483) (1,483) (1,6370)	
UBS Global Asset Namagement By 3,200 4,28 UBS Scornites Australia Ltd Boy 3,712 (5,166) UBS Securities Australia Ltd Buy 47,377 (5,166) UBS Securities Australia Ltd Buy 47,377 (6,1022a) UBS Securities Australia Ltd Buy 917,483 (6,1020a) UBS Securities Australia Ltd Buy 73,203 (6,1070b) UBS Securities Australia Ltd Buy 73,204 (6,1070b) UBS Securities Australia Ltd Buy 919,483 (6,1070b) UBS Securities Australia Ltd Buy 919,400 (10,000b) UBS Securities Australia Ltd Buy 919,400 (10,000b) UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,506 2,506 UBS Securit	 	bal Asset Management urities Australie Ltd urities Australie Ltd London Branch London Branch London Branch Linites Australie Ltd urities Australie Ltd urities Australie Ltd urities Australie Ltd urities Australie Ltd urities Australie Ltd urities Australie Ltd urities Australie Ltd urities Australie Ltd urities Australie Ltd urities Australie Ltd	Buy Buy Buy Buy Buy Buy Buy Sell Sell Sell Buy Buy Buy Buy	3,926 47,372 211 211 211,134 919,483 183,994 183,994 183,994 19,340 91,100 91,400 91,400 91,400 91,400 91,400 91,400 91,400 91,400 91,400	(5,166) 23 23 24 (100,524) 100,524 (4,483) (4,483) (20,060) (16,377)	
UBS Securities Australia Ltd Self 47,372 (5,166) UBS Securities Australia Ltd Buy 211 23 UBS Securities Australia Ltd Buy 217 173 BBS Act London Branch Sell 917,174 (100,224) UBS Securities Australia Ltd Buy 919,483 100,224 UBS Securities Australia Ltd Sell 13,249 (20,0060) UBS Securities Australia Ltd Buy 919,482 100,002 UBS Securities Australia Ltd Buy 919,482 10,000 UBS Securities Australia Ltd Buy 91,400 (1,000) UBS Securities Australia Ltd Buy 2,60 2,73 UBS Securities Australia Ltd Buy 2,60 2,73 UBS Securities Australia Ltd Buy 2,60 1,73 UBS Securities Australia Ltd Buy 2,60 2,73 UBS Securities Australia Ltd Buy 2,60 2,73 UBS Securities Australia Ltd Buy 2,60 2,73 UBS Securities Australia Ltd	 	unties Australia Ltd unties Australia Ltd unties Australia Ltd London Branch London Branch unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd	Self Buy Self Buy Self Self Self Self Self Self Self Self	47,372 211 211 211 917,174 919,483 40,021 183,994 110,934 410,484 91,400 91,400 91,400 247,281 280,161 14,573	(5,16) 23 23 (100,524) 100,524 (4,483) (2,060) (16,377)	
URS Securities Australia Ltd Boy 2.72 3.71 URS Securities Australia Ltd Blow 2.11 2.3 URS Securities Australia Ltd Sell 917,483 100,524,1 URS Securities Australia Ltd Sell 40,284 100,524,1 URS Securities Australia Ltd Bloy 91,483 100,524,1 URS Securities Australia Ltd Bloy 73,203 6,430 URS Securities Australia Ltd Bloy 73,203 6,630 URS Securities Australia Ltd Bloy 73,203 6,637 URS Securities Australia Ltd Bloy 91,100 1,000 URS Securities Australia Ltd Bloy 2,566 2,715 URS Securities Australia Ltd Bloy 2,566 2,716 URS Securities Australia Ltd Bloy 2,566 2,716 URS Securities		urities Australia Ltd rivities Australia Ltd Condon Branch Condon Branch Condon Branch Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd Trifes Australia Ltd	Buy Buy Buy Sell Sell Sell Sell Buy Buy Buy	431 431 431 40,921 73,203 183,994 183,994 183,994 193,994 10,493 91,100 91,100 91,100 91,100 91,100 14,573 14,573	(100,524) (100,524) (100,524) (4,48) (4,630) (20,060)	
Obes Securities Australia Ltd Buy 2.13 4.21 4.22 4.21 4.23 <	1 1 1 1 1 1 1 1 1	urities Australia Ltd London Branch London Branch London Branch London Branch Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd Linies Australia Ltd	Buy Buy Sell Buy Sell Buy Sell Buy Buy	217,724 917,724 919,483 40,927 183,994 149,344 10,493 91,100 91,100 91,100 91,100 91,400 91,400 91,400 14,573 14,573	(100,524) (100,524) (4,483) (20,060) (16,377)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 917,174 (100,524) UBS AG London Barach Buy 917,174 (100,524) UBS AG London Barach Buy 919,483 100,224 UBS AG London Barach Buy 40,232 (14,882) UBS Ageourities Australia Ltd Buy 73,203 6,030 UBS Securities Australia Ltd Buy 719,483 10,000 UBS Securities Australia Ltd Buy 91,400 10,000 UBS Securities Australia Ltd Buy 2,506 2,757 UBS Securities Australia Ltd	1 1 1 1 1 1 1 1	London Branch London Branch London Branch Lordon Branch Lordon Branch Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd Linites Australia Ltd	Sell Buy Sell Sell Sell Sell Buy Buy	917,174 919,483 40,521 73,208 183,994 110,934 410,934 410,934 91,100 91,100 247,981 247,981 247,981 247,981 145,73	(100,524) 100,524 (4,483) 8,030 (20,060)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Ones Are controls Series 917,174 10,023-04 UBS Securities Australia Ltd Sell 4,092 1,002-05 UBS Securities Australia Ltd Sell 73,003 6,030 UBS Securities Australia Ltd Sell 183,944 4,0420 UBS Securities Australia Ltd Buy 73,003 6,030 UBS Securities Australia Ltd Buy 410,432 1,000 UBS Securities Australia Ltd Buy 91,400 (10,000) UBS Securities Australia Ltd Buy 91,400 (10,000) UBS Securities Australia Ltd Buy 22,508 1,534 UBS Securities Australia Ltd Buy 2,508 1,94 UBS Securities Australia Ltd Buy 2,508 1,00 UBS Securities Australia Ltd Buy 2,508 2,306 UBS Securities Australia	1 1 1 1 1 1 1	London fearth London Barach London Barach urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Sell Buy Sell Sell Buy Buy Buy	917,174 40,221 73,203 183,994 149,344 10,494 91,100 91,100 91,100 247,981 287,981 14,573	(100,524) 100,524 (4,483) 8,030 (20,060)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Astratials Ltd Bely 197,485 100,324 UBS Securities Astratials Ltd Buy 40,321 (4,483) UBS Securities Astratials Ltd Buy 73,203 (8,030) UBS Securities Astralials Ltd Buy 1100 44,528 UBS Securities Astralials Ltd Buy 91,100 10,000 UBS Securities Astralials Ltd Buy 91,100 10,000 UBS Securities Astralials Ltd Buy 91,400 10,000 UBS Securities Astralials Ltd Buy 2,506 273 UBS Securities Astralials Ltd Buy 2,506 273 UBS Securities Astralials Ltd Buy 2,506 274 UBS Securities Astralials Ltd Buy 4,61 7 UBS Securities Astralials Ltd Buy 2,306 2,306 UBS Securities A	1 1 1 1 1 1	urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Buy Sell Sell Buy Buy	919,483 40,921 183,994 149,344 410,494 919,489 91,400 91,400 247,981 14,573 14,573	(4,483) (4,483) 8,030 (20,060)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 73,201 (4,030 UBS Securities Australia Ltd Sell 183,994 (2,020) UBS Securities Australia Ltd Buy 410,4934 (4,032) UBS Securities Australia Ltd Buy 410,4934 (4,022) UBS Securities Australia Ltd Buy 410,4934 (16,000) UBS Securities Australia Ltd Buy 41,133 (10,000) UBS Securities Australia Ltd Buy 24,138 (10,000) UBS Securities Australia Ltd Buy 2,47,581 (27,157) UBS Securities Australia Ltd Buy 2,67 237 UBS Securities Australia Ltd Buy 2,67 241 UBS Securities Australia Ltd Buy 2,64 7 UBS Securities Australia Ltd Buy 2,794 44 UBS Securities Australia Ltd Buy 2,794 2,74 UBS Securities Australia Ltd Buy 2,794 2,74 UBS Securities Australia Ltd Buy 2,794 2,74 UBS Securities Australi	1 1 1 1 1	uriles Australia t.c. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d. uriles Australia t.d.	Buy Sell Sell Buy Buy	40,927 73,208 183,994 110,494 410,494 91,400 247,981 280,161 180,781 14,573 14,573	(4,483) 8,030 (20,060)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 7.3,03 UBS Securities Australia Ltd Sell 183,944 (16,377) UBS Securities Australia Ltd Buy 40,493.44 (16,377) UBS Securities Australia Ltd Buy 40,493.44 (16,377) UBS Securities Australia Ltd Buy 919,483 10,622.8 UBS Securities Australia Ltd Buy 919,483 10,622.8 UBS Securities Australia Ltd Buy 919,483 1,615.9 UBS Securities Australia Ltd Buy 27,261 1,534 UBS Securities Australia Ltd Buy 2,566 2,77 UBS Securities Australia Ltd Buy 2,64 7 UBS Securities Australia Ltd Buy 2,64 7 UBS Securities Australia Ltd Buy 2,784 307 UBS Securities Australia Ltd Buy 2,784 307 UBS Securities Australia Ltd Buy 3,144 3,14 UBS Securities Australia Ltd Buy 2,267 2,267 UBS Securities Australia Ltd	1111	rrities Australia Ltd rrities Australia Ltd rrities Australia Ltd rrities Australia Ltd rrities Australia Ltd rrities Australia Ltd rrities Australia Ltd rrities Australia Ltd	Buy Sell Buy Buy	73,203 183,994 10,493 91,100 91,100 91,100 247,981 287,981 14,573 14,573	(20,060)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Sell 149.344 (EX)00 UBS Securities Australia Ltd Buy 410.494 44.928 UBS Securities Australia Ltd Buy 410.494 44.938 UBS Securities Australia Ltd Buy 410.493 44.938 UBS Securities Australia Ltd Buy 91,100 10.000 UBS Securities Australia Ltd Buy 91,100 10.000 UBS Securities Australia Ltd Buy 2,506 27.33 UBS Securities Australia Ltd Buy 2,506 27.33 UBS Securities Australia Ltd Buy 2,506 27.34 UBS Securities Australia Ltd Buy 2,506 2.506 UBS Securities Australia L	111	rifies Australia tro rifies Australia tro rifies Australia tro rifies Australia tro rifies Australia tro rifies Australia tro rifies Australia tro rifies Australia tro	Sell Buy Buy Buy	183,594 110,344 110,048 91,100 91,100 91,400 247,581 247,581 14,573	(20,060)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Swill 449,344 4(16,377) UBS Securities Australia Ltd Buy 410,484 4(16,377) UBS Securities Australia Ltd Buy 919,482 100,522 UBS Securities Australia Ltd Buy 919,482 100,522 UBS Securities Australia Ltd Buy 247,281 10,000 UBS Securities Australia Ltd Buy 1,4713 1,615 UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,506 3,71 UBS Securities Australia Ltd Buy 2,606 3,71 UBS Securities Australia Ltd Buy 2,78 4 UBS Securities Australia Ltd <td< td=""><td>i 1</td><td>unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd</td><td>Buy Buy Buy</td><td>149,344 410,494 919,483 91,100 91,400 247,581 280,161 14,573</td><td>(16,377)</td><td>Ordinary Ordinary Ordinary Ordinary Ordinary</td></td<>	i 1	unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd	Buy Buy Buy	149,344 410,494 919,483 91,100 91,400 247,581 280,161 14,573	(16,377)	Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 910,483 44,928 UBS Securities Australia Ltd Buy 91,000 10,000 UBS Securities Australia Ltd Sell 21,000 10,000 UBS Securities Australia Ltd Buy 280,161 30,710 UBS Securities Australia Ltd Buy 280,161 30,710 UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,206 273 UBS Securities Australia Ltd Buy 2,207 247 UBS Securities Australia Ltd Buy 2,206 2,27 UBS Securities Australia Ltd Buy 2,267 247 UBS Securities Australia Ltd Buy 2,267 247 UBS Securities Australia Ltd Buy 2,267 247 UBS Securities Australia Ltd Buy 5,88 3,86 UBS Securities Australia Ltd Buy 2,366 2,80 UBS Securities Australia Ltd Buy 2,366 2,80 UBS Securities Australia Ltd		untes Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Buy Buy Buy	410,494 919,483 91,100 91,400 247,981 280,161 14,573		Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 919,483 100,524 UBS Securities Australia Ltd Buy 91,100 10,000 UBS Securities Australia Ltd Sell 247,981 (07,000) UBS Securities Australia Ltd Buy 28,015 1,594 UBS Securities Australia Ltd Buy 2,807 441 UBS Securities Australia Ltd Buy 2,267 247 UBS Securities Australia Ltd Buy 2,366 273 UBS Securities Australia Ltd Buy 2,366 273 UBS Securities Australia Ltd Buy 2,366 278 UBS Securities Australia Ltd Buy 2,366 2,366 UBS Securities Australia Ltd Buy 2,366 2,366 UBS Securities Australia Ltd Bu		urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Buy Buy	919,483 91,100 91,400 247,981 280,161 14,573	44,928	Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 91,100 10,000 UBS Securities Australia Ltd Sell 31,400 (70,000) UBS Securities Australia Ltd Buy 247,981 (27,157) UBS Securities Australia Ltd Buy 2,505 273 UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,506 2,73 UBS Securities Australia Ltd Buy 2,607 247 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 31,49 304 UBS Securities Australia Ltd Buy 2,794 37 UBS Securities Australia Ltd Buy 2,794	ı	urities Australia Ltd urities Australia Ltd urities Australia Ltd	Buy	91,100 91,400 247,981 280,161 14,573	100,524	Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Sell 247,981 (10,000) UBS Securities Australia Ltd Buy 28,01 61 30,710 UBS Global Asset Management Buy 28,01 61 30,710 UBS Global Asset Management Buy 28,01 61 30,710 UBS Global Asset Management Australia Ltd Buy 2,506 27.37 UBS Securities Australia Ltd Buy 2,506 2.73 UBS Securities Australia Ltd Buy 2,506 2.73 UBS Securities Australia Ltd Buy 46 7 UBS Securities Australia Ltd Buy 46 7 UBS Securities Australia Ltd Buy 3,149 3,6 UBS Securities Australia Ltd Buy 3,149 3,6 UBS Securities Australia Ltd Buy 2,794 4 UBS Securities Australia Ltd Buy<	- 1	urities Australia Ltd urities Australia Ltd		91,400 247,981 280,161 14,573	10,000	Ordinary Ordinary
UBS Securities Australia Ltd Sell 247,981 (27,157) UBS Securities Australia Ltd Buy 280,161 30,710 UBS Global Asset Management Buy 280,161 30,710 UBS Global Asset Management Buy 14,713 1,615 UBS Securities Australia Ltd Buy 44,11 48 UBS Securities Australia Ltd Buy 2,267 247 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 57 4 UBS Securities Australia Ltd Buy 57 4 UBS Securities Australia Ltd Buy 2,366 2,366 UBS Securities Australia Ltd Buy 2,366 2,47 UBS Securities Australia Ltd Buy 2,366 2,46 UBS Securities Australia Ltd Buy 2,366 2,366 UBS Securities Australia Ltd Buy 2,366 </td <td>_</td> <td>urities Australia Ltd</td> <td>Seil</td> <td>247,981 280,161 14,573</td> <td>(10,000)</td> <td>Ordinary</td>	_	urities Australia Ltd	Seil	247,981 280,161 14,573	(10,000)	Ordinary
URS Securities Australia Ltd Buy 280,151 30,710 UBS Securities Australia Ltd Buy 1,534 1,534 UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,607 2,41 UBS Securities Australia Ltd Buy 6,4 7 UBS Securities Australia Ltd Buy 4,6 5 UBS Securities Australia Ltd Buy 4,6 5 UBS Securities Australia Ltd Buy 4,6 5 UBS Securities Australia Ltd Buy 3,149 3,0 UBS Securities Australia Ltd Buy 3,149 3,0 UBS Securities Australia Ltd Buy 3,149 3,0 UBS Securities Australia Ltd Buy 2,794 4 UBS Securities Australia Ltd Buy 2,794 4 UBS Securities Australia Ltd Buy 2,794 4 UBS Securities Australia Ltd Buy 2,794 <			Sell	280,161	(27,157)	
UBS Global Asset Management Buy 14,573 1,594 UBS Global Asset Management Buy 14,713 1,615 UBS Securities Australia Ltd Buy 2,506 247 UBS Securities Australia Ltd Buy 2,607 247 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 65 6 UBS Securities Australia Ltd Buy 3,149 3,46 UBS Securities Australia Ltd Buy 2,784 3,07	Г	urities Australia Ltd	Buy	14,573	30,710	Ordinary
UBS Wealth Management Australia Ltd Buy 14,713 1,615 UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 441 448 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 31.49 346 UBS Securities Australia Ltd Buy 374 37 UBS Securities Australia Ltd Buy 2.794 30 UBS Securities Australia Ltd Buy 2.794 36 UBS Securities Australia Ltd Buy 2.794 37 UBS Securities Australia Ltd Buy 2.306 2.86 UBS Securities Australia Ltd Buy 2.306 2.86	Г	bal Asset Management	Suv		1.594	Ordinary
UBS Securities Australia Ltd Buy 2,506 273 UBS Securities Australia Ltd Buy 2,506 2,73 UBS Securities Australia Ltd Buy 2,64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 9 11 UBS Securities Australia Ltd Buy 3,149 2 UBS Securities Australia Ltd Buy 3,149 1 UBS Securities Australia Ltd Buy 3,149 3,149 UBS Securities Australia Ltd Buy 2,794 4	Т	alth Management Australia I td	Riv	14 713	1 615	Ordinary
UBS Securities Australia Ltd Buy 2.267 247 UBS Securities Australia Ltd Buy 2.267 247 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 5149 346 UBS Securities Australia Ltd Buy 109 11 UBS Securities Australia Ltd Buy 3149 346 UBS Securities Australia Ltd Buy 36 6 UBS Securities Australia Ltd Buy 2,365 2,60 UBS Securities Australia Ltd Buy 2,365 328 UBS Securities Australia Ltd Buy 2,365 2,60 UBS Securities Australia Ltd Buy 2,365 2,60 UBS Securities Australia Ltd Buy 2,360 2,50 <	1	urities Australia Ltd	Rin	2.506	273	Ordinary
UBS Securities Australia Ltd Buy 2.267 247 UBS Securities Australia Ltd Buy 6.4 7 UBS Securities Australia Ltd Buy 19 1 UBS Securities Australia Ltd Buy 3.149 3.6 UBS Securities Australia Ltd Buy 3.149 3.6 UBS Securities Australia Ltd Buy 3.6 4 UBS Securities Australia Ltd Buy 3.6 4 UBS Securities Australia Ltd Buy 2.794 3.07 UBS Securities Australia Ltd Buy 2.794 4 UBS Securities Australia Ltd Buy 2.794 4 UBS Securities Australia Ltd Buy 2.794 4 UBS Securities Australia Ltd Buy 2.794 5	Ť	ritiac Australia i td	Riv	441	48	Ordinary
UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 9 11 UBS Securities Australia Ltd Buy 190 11 UBS Securities Australia Ltd Buy 31.49 346 UBS Securities Australia Ltd Buy 37.94 307 UBS Securities Australia Ltd Buy 2.794 4 UBS Securities Australia Ltd Buy 2.365 6 UBS Securities Australia Ltd Buy 2.366 6 UBS Securities Australia Ltd Buy 2.360 2.56	۲	Latin Australia 144	200	736.0	74.6	Cicinaly
UBS Securities Australia Ltd Buy 544 139 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 191 21 UBS Securities Australia Ltd Buy 100 11 UBS Securities Australia Ltd Buy 3,149 3,46 UBS Securities Australia Ltd Buy 3,149 3,46 UBS Securities Australia Ltd Buy 3,149 3,46 UBS Securities Australia Ltd Buy 2,794 3,07 UBS Securities Australia Ltd Buy 2,985 3,08 UBS Securities Australia Ltd Buy 2,985 3,08 UBS Securities Australia Ltd Buy 2,985 3,08 UBS Securities Australia Ltd Buy 2,386 4 UBS Securities Australia Ltd Buy 2,380 4 UBS Securities Australia Ltd Buy 2,380 5 </td <td>Ť</td> <td>mitting Australia 144</td> <td>sol.</td> <td>2,207</td> <td>(54)</td> <td>Ordinary</td>	Ť	mitting Australia 144	sol.	2,207	(54)	Ordinary
UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 191 11 UBS Securities Australia Ltd Buy 191 11 UBS Securities Australia Ltd Buy 3149 346 UBS Securities Australia Ltd Buy 37 4 UBS Securities Australia Ltd Buy 37 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,794 307 UBS Securities Australia Ltd Buy 2,794 4 UBS Securities Australia Ltd Buy 2,794 4 UBS Securities Australia Ltd Buy 2,366 4 UBS Securities Australia Ltd Buy 2,366 4 UBS Securities Australia Ltd Buy 2,366 4 UBS Securities Australia Ltd Buy 2,360 2,56	Ť	mines According to a	136	100	(CC)	Gunary
UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 46 7 UBS Securities Australia Ltd Buy 191 21 UBS Securities Australia Ltd Buy 190 11 UBS Securities Australia Ltd Buy 100 11 UBS Securities Australia Ltd Buy 31.49 346 UBS Securities Australia Ltd Buy 27.94 30.7 UBS Securities Australia Ltd Buy 2.794 4 UBS Securities Australia Ltd Buy 2.365 6 UBS Securities Australia Ltd Buy 2.365 6 UBS Securities Australia Ltd Buy 2.330 2.56 UBS Securities Australia Ltd Buy 2.330 2.56 UBS Securities Australia Ltd Buy 2.330 2.	+	Julies Australia Lid	auy	40	\	Ordinary
UBS Securities Australia Ltd Buy 54 5 UBS Securities Australia Ltd Buy 51 6 UBS Securities Australia Ltd Buy 191 21 UBS Securities Australia Ltd Buy 100 11 UBS Securities Australia Ltd Buy 3,149 3,6 UBS Securities Australia Ltd Buy 2,794 307 UBS Securities Australia Ltd Buy 2,965 366 UBS Securities Australia Ltd Buy 2,965 374 UBS Securities Australia Ltd Buy 2,366 4 UBS Securities Australia Ltd Buy 2,366 4 UBS Securities Australia Ltd Buy 2,330 2,66 UBS Securities Australia Ltd Buy 2,330 2,6 UBS Securities Australia Ltd Buy 2,330 2,	7	Jrities Australia Ltd	griý	64	7	Ordinary
UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 191 71 UBS Securities Australia Ltd Buy 9 11 UBS Securities Australia Ltd Buy 3,149 346 UBS Securities Australia Ltd Buy 36 6 UBS Securities Australia Ltd Buy 2,734 307 UBS Securities Australia Ltd Buy 2,736 200 UBS Securities Australia Ltd Buy 2,736 200 UBS Securities Australia Ltd Buy 2,736 4 UBS Securities Australia Ltd Buy 2,736 4 UBS Securities Australia Ltd Buy 2,736 4 UBS Securities Australia Ltd Buy 56 6 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 64 6 <tr< td=""><td>7</td><td>urities Australia Ltd</td><td>Buy</td><td>46</td><td>S</td><td>Ordinary</td></tr<>	7	urities Australia Ltd	Buy	46	S	Ordinary
UBS Securities Australia Ltd Buy 191 21 UBS Securities Australia Ltd Buy 100 11 UBS Securities Australia Ltd Buy 3,149 346 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 2,965 200 UBS Securities Australia Ltd Buy 2,965 328 UBS Securities Australia Ltd Buy 2,965 328 UBS Securities Australia Ltd Buy 6 4 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 53 6 UBS Securities Australia Ltd Buy 53 6 UBS Securities Australia Ltd Buy 53 6 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7	_	urities Australia Ltd	Buy	55	9	Ordinary
URB Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 3.149 3.46 UBS Securities Australia Ltd Buy 3.149 3.46 UBS Securities Australia Ltd Buy 3.6 4 UBS Securities Australia Ltd Buy 2.794 30.7 UBS Securities Australia Ltd Buy 2.796 2.60 UBS Securities Australia Ltd Buy 2.796 2.6 UBS Securities Australia Ltd Buy 2.796 5 UBS Securities Australia Ltd Buy 2.6 6 UBS Securities Australia Ltd Buy 2.7 4 UBS Securities Australia Ltd Buy 2.7 4 UBS Securities Australia Ltd Buy 2.7 3 UBS Securities Australia Ltd Buy 2.7 4	_	urities Australia Ltd	Buy	191	21	Ordinary
UBS Securities Australia Ltd Buy 100 11 UBS Securities Australia Ltd Buy 3149 346 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 5794 307 UBS Securities Australia Ltd Buy 2,794 307 UBS Securities Australia Ltd Buy 2,796 260 UBS Securities Australia Ltd Buy 2,366 250 UBS Securities Australia Ltd Buy 2,366 260 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 2,330 256	ř	urities Australia Ltd	Buy	6		Ordinary
UBS Securities Australia Ltd Buy 3,149 346 UBS Securities Australia Ltd Buy 18 2 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,794 307 UBS Securities Australia Ltd Buy 2,985 328 UBS Securities Australia Ltd Buy 2,985 328 UBS Securities Australia Ltd Buy 5 6 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 5 6 UBS Securities Australia Ltd Buy 5 6 UBS Securities Australia Ltd Buy 5 6 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 2,330 2 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 2,330 2 UBS Securities Australia Ltd Buy 2,330 2	f	urities Australia I td	Risc	100	-	Ordinany
UBS Securities Australia Ltd Buy 3,143 340 UBS Securities Australia Ltd Buy 5,73 6 UBS Securities Australia Ltd Buy 2,734 307 UBS Securities Australia Ltd Buy 2,734 307 UBS Securities Australia Ltd Buy 2,736 260 UBS Securities Australia Ltd Buy 2,736 4 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 56 6 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 2,330 2,7 UBS Securities Australia Ltd Buy 2,330 2,6 UBS Securities Australia Ltd Buy 2,330 2,6	۲	rition Australia 14d	, in a	2140	740	Oldinal y
UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 2,794 307 UBS Securities Australia Ltd Buy 2,796 307 UBS Securities Australia Ltd Buy 2,985 328 UBS Securities Australia Ltd Buy 4 4 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 4 6 UBS Securities Australia Ltd Buy 4 6 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 230 256 UBS Se	+	annes Australia cru	Duy	5,149	340	Orginary
UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,734 307 UBS Securities Australia Ltd Buy 2,736 250 UBS Securities Australia Ltd Buy 2,985 378 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 56 4 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 6,4 7 UBS Securities Australia Ltd Buy 6,4 7 UBS Securities Australia Ltd Buy 6,4 6 UBS Securities Australia Ltd Buy 6,4 6 UBS Securities Australia Ltd Buy 2,7 3 UBS Securities Australia Ltd Buy 2,330 2,6 UBS Securities Australia Ltd Buy 2,330 2,6 UBS Securities Australia Ltd Buy 2,330 2,6	7	Jrities Australia Ltd	Buy	18	7	Ordinary
UBS Securities Australia Ltd Buy 35 4 UBS Securities Australia Ltd Buy 2,734 307 UBS Securities Australia Ltd Buy 2,366 328 UBS Securities Australia Ltd Buy 2,985 328 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 230 256 UBS Securities Australia Ltd Buy 230 26 UBS Securities Australia Ltd Buy 230 26 UBS Securities Australia Ltd Buy 230 26 <td< td=""><td>_</td><td>urities Australia Ltd</td><td>Buy</td><td>55</td><td>9</td><td>Ordinary</td></td<>	_	urities Australia Ltd	Buy	55	9	Ordinary
UBS Securities Australia Ltd Buy 2.794 307 UBS Securities Australia Ltd Buy 2,366 260 UBS Securities Australia Ltd Buy 2,985 128 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 55 4 UBS Securities Australia Ltd Buy 56 4 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 4 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 2,330 26 UBS Securities Australia Ltd Buy 2,330 2,14	- 1	urities Australia Ltd	Buy	36	4	Ordinary
URS Securities Australia Ltd Buy 2,366 260 UBS Securities Australia Ltd Buy 1,985 328 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 2,330 2,56 <t< td=""><td></td><td>urities Australia Ltd</td><td>Buy</td><td>2,794</td><td>307</td><td>Ordinary</td></t<>		urities Australia Ltd	Buy	2,794	307	Ordinary
UBS Securities Australia Ltd Buy 2,965 328 UBS Securities Australia Ltd Buy 109 12 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 44 6 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 2,330 2,66 UBS Securities Australia Ltd Buy 2,330 2,66 UBS Securities Australia Ltd Buy 2,330 2,66 UBS Securities Australia Ltd Buy 3,9 1,14	ı	urities Australia Ltd	Buy	2 366	260	Ordinary
UBS Securities Australia Ltd Buy 109 12 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 5.330 2.66 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 2,330 2,6 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 6 5 UBS Securities Australia Ltd Buy 6 5 UBS Securities Australia Ltd Buy 6 7 UBS Secu		irities Australia Itd	Riv	7 985	328	Ordinan
UBS Securities Australia Ltd Buy 46 12 UBS Securities Australia Ltd Buy 46 6 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 65 6 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 23 24 UBS Securities Australia Ltd Buy 2330 26 UBS Securities Australia Ltd Buy 2330 230 UBS Securities Australia Ltd Buy 230 1 U	1	total cilculation of the	600	200,	25.	or all all
UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,730 2,76 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 6 1 UBS Securities Australia Ltd Buy 6 5 UBS Securities Australia Ltd Buy 1,192 5 UBS Secu	7	Illies Australia Liu	had	601	7	Ordinary
UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 23 4 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 2,330 2,66 UBS Securities Australia Ltd Buy 2,330 2,66 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 933 108	7	Intles Australia Ltd	Buy	46	2	Ordinary
UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 25 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 46 6 UBS Securities Australia Ltd Buy 46 6 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108	_	irities Australia Ltd	Buy	55	9	Ordinary
UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 67 6 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 6 5 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 983 108	_	irities Australia Ltd	Buy	36	4	Ordinary
UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 65 6 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 2,330 2,6 UBS Securities Australia Ltd Buy 2,330 2,6 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 1,132 131 UBS Securities Australia Ltd Buy 983 108	Ť	irities Australia Ltd	Buy	36	4	Ordinary
UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 64 7 UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 1,192 5 UBS Securities Australia Ltd Buy 1,92 5 UBS Securities Australia Ltd Buy 1,92 3	Ť	rities Australia I td	Rin	2 330	756	Ordinary
UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 2,330 2,66 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108	✝¯	reties Australia I to	Sin	20013		Ordinary Strains
UBS Securities Australia Ltd Buy 55 6 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 2,330 5.6 UBS Securities Australia Ltd Buy 39 1 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 1,88 108	Ť	Tri I I I I I I I I I I I I I I I I I I I	(ha	t i		O'CILIBIY
UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 2,56 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108	۲	irities Australia Ltd	euy 6	cc.	ام	Ordinary
UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 36 4 UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108	1	urties Australia Ltd	Buy	4P	5	Ordinary
UBS Securities Australia Ltd Buy 3.6 4 UBS Securities Australia Ltd Buy 2,330 2.56 UBS Securities Australia Ltd Buy 6 7 UBS Securities Australia Ltd Buy 4 5 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108		irities Australia Ltd	Buy	27	3	Ordinary
UBS Securities Australia Ltd Buy 2,330 256 UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108		rrities Australia Ltd	Buy	36	4	Ordinary
UBS Securities Australia Ltd Buy 9 1 UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108	Г	irities Australia Ltd	Buy	2,330	256	Ordinary
UBS Securities Australia Ltd Buy 46 5 UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108		irities Australia Ltd	Buv	ı	-	Ordinary
UBS Securities Australia Ltd Buy 1,192 131 UBS Securities Australia Ltd Buy 983 108	1	irities Australia Ltd	Bity	46	4	Ordinary
UBS Securities Australia Ltd 80y 983 108		iritios Δixtralia Itd	Rin	1 192	131	Ordinary
UBS Securities Australia Ltd Buy 108	1		, and	200	2 6	Cicinaly
	╗	irities Australia Ltd	Buy	983	ΣΩL	Ordinary

(1)					
16-May-14	LIBC Global Asset Management	Buy	38,861	4,213	Ordinary
6-May-14	UBS Global Asset Management	Bity	320,020	34 017	Ordinary
16-May-14	I IRS Global Asset Management	Biry	25.215	110,40	Cidinary
16-May-14	1 (BC Global Acces Management	500	000.00	2,733	Ordinary
17	1	and a	50,389	2,210	Ordinary
O-way-14	Obsignation Asset Management	gnà	7//0/	1,6/1	Ordinary
p-Ividy-14	Obsidional Asset imanagement	euy	196,836	21,335	Ordinary
6-May-14	UBS Global Asset Management	Buy	249,839	27,080	Ordinary
6-May-14	UBS Global Asset Management	Buy	698,847	75,748	Ordinary
6-May-14	UBS Global Asset Management	Buy	185,591	9,279	Ordinary
6-May-14	UBS Global Asset Management	Buy	166.818	18 085	Ordinary
6-May-14	UBS Global Asset Management	Buy	141 212	15 309	Ordinary
6-May-14	I BS Global Asset Management	Birk	250 730	00100	in in o
		fno.	907,600	59,109	Gamary
-May-14	Asset	Buy	721,107	76,223	Ordinary
6-May-14	loba	Buy	4,421,568	480,595	Ordinary
-May-14	UBS Global Asset Management	Buy	1,353,695	147,167	Ordinary
16-May-14	UBS Global Asset Management	Buy	105,968	11,518	Ordinary
6-May-14	UBS Global Asset Management	S	85 691	9314	Ordinary
-May-14	UBS Global Asset Management	Biry	207 433	37 370	ording.
16-May-14	II IRS Global Asset Management	2 6	100, 100	120,00	Cicinal
16-May-14	I DC Global Asset Management		167,120	176,50	Colling
the latest	CDS CICLOSE Master Management	puy	ccu,ucu,1	14,134	Ordinary
D-May-14	UBS Global Asset Management	suy	2,937,181	319,252	Ordinary
5-May-14	UBS Global Asset Management	Buy	593,497	64,522	Ordinary
16-May-14	UBS Global Asset Management	Buy	2,236	243	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Buy	11,469	1.248	Ordinary
16-May-14	UBS Wealth Management Australia Ltd	3	12,257	(1 335)	Ordinary
6-May-14	UBS Wealth Management Australia 11d	و	514	(55)	Ordinan
May 14	I IBC M/solth Management Authority			(00)	O CI III
	CO Weath was agent and a con-	ing.	14,490	//5'1	Orginary
-Ividy-14	Obs wealth wanagement Australia Ltd	Sell	3,0/6	(335)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	19,973	(2,171)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	1,122	(122)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	117,852	(12,810)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	111,614	(12 132)	Ordinary
-Mav-14	UBS Securities Australia I1d	ay	13 202	(1 435)	Ordinan
K-May-17	I IBC Counties Australia 14d	100	202,01	(001.)	Ordinally Ordinally
6-May-14	L'BC Coourities Australia 14d	100	10,01	(000)	Cicialy
,	Control of the contro	100	3,102	(966)	Ordinary
O-May 14	ODS Securities Abstratia Ltd	nac J	17,140	(1,863)	Ordinary
o-May-14	UBS Securities Australia Ltd	Sell	5,005	(544)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	4,223	(428)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	1,334	(145)	Ordinary
6-May-14	UBS Securities Australia Ltd	Seli	46	(2)	Ordinary
6-May-14	UBS Sequities Australia I td	[lay	V08 60	(10 000)	Ording.
16-May-14	LIBO Convision Australia 144	100	460,65	(10,030)	Oldinary
-ividy-	ODD DECEMBES AUSTRAIN LIU	ner c	102,378	(11,128)	Ordinary
b-May-14	Obs securities Australia Ltd	Sell	5,686	(618)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	116,012	(12,610)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	94,788	(10,303)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	3.763	(404)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	10 939	(1.189)	Ordinary
6-May-14	I IRS Securities Australia I to	Cell	300 c	(2011)	Ordinary
May 14	Por Committee Australia 144	11-3	00000	(166)	Oldinary
	ביים וויים ביים וויים ביים ביים	000	12,000	(5,5,1)	Orginary
-May-14	UBS Securities Australia Ltd	Seli	1,388	(120)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	1,388	(120)	Ordinary
16-May-14	UBS Securities Australia Ltd	Self	3,608	(360)	Ordinary
-Mav-14	UBS Securities Australia Ltd	llay	148	(15)	Ordinan
May 14	I IBS Securities Australia I to	100	6.101	(01)	(in in in in in in in in in in in in in i
1 1 1 1 1	One Commerce According to	100	9,103	(000)	Ordinary
-Mdy-14	UBS Securities Australia Lto	Seil	8,751	(946)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	1,388	(150)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	3,580	(387)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	10,379	(1,122)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	574	(62)	Ordinary
6-Mav-14	UBS Securities Australia Ltd	Sel	814	(88)	Ordinan
6-May-14	I IBS Societies Australia 14d	100	1 501	(475)	Old fall
	DE Commission American	100	166,1	(7/1)	Ordinary
b-May-14	UBS Securities Australia Ltd	Sell	5,458	(230)	Ordinary
6-May-14	UBS Securities Australia Ltd	Zell	18.500	1000	
				(2,000)	Ordinary
16-May-14	UBS Securities Australia Ltd	las.	2.932	(317)	Ordinary

1 1 1 1	UBS Securities Australia Ltd	Buy	1,320	145	Ordinary
	LIDS Con mitton Auctoria 14d				
	Opp permittee Australia Liu	Buy	36	4	Ordinary
ı	UBS Securities Australia Ltd	Buy	1,629	179	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	846	93	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	346	88	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	355	39	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	309	34	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	36	4	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	36	4	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	46	5	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	27	3	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	508	23	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	3,640	400	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	18,200	2,000	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	7,862	864	Ordinary
	UBS Securities Australia Ltd	Buy	12,494	1,373	Ordinary
ı	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	783	98	Ordinary
1	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	2,366	790	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	2,248	247	Ordinary
5-May-14	UBS Securties Australia Ltd	Buy	159,569	17,535	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	22,213	2,441	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	14,123	1,552	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	9,100	1,000	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	12,403	1,363	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	6,852	753	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	7,862	864	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	2,503	275	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	32,851	3,610	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	7,862	864	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	27,300	3,000	Ordinary
ı	UBS Securities Australia Ltd	Buy	2,730	300	Ordinary
	UBS Securities Australia Ltd	Buy	48,212	5,298	Ordinary
15-May-14	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
	UBS Securities Australia Ltd	Buy	289	70	Ordinary
5-May-14	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
1	UBS Securities Australia Ltd	Buy	2.366	260	Ordinary
1	UBS Securities Australia Ltd	Buv	3.149	346	Ordinary
15-May-14	UBS Securities Australia Ltd	Buv	3.149	346	Ordinary
ı	UBS Securities Australia Ltd	8uv	3,149	346	Ordinary
ı	UBS Securities Australia Ltd	Buy	2,830	311	Ordinary
15-May-14	UBS Securities Australia I td	Buy	7 862	864	Ordinary
1	IBS Securities Australia I to	Bin	18.45	acu c	Ordinary
5-May-14	I IBS Societies Australia Ltd	Co. d	300 /	2,020	Ordinary
	Obsider Australia Liu	eug Pers	4,035	450	Ordinary
1	DDS Secondies Australia Ltd	Series Series	318	32	Ordinary
T	UBS Securities Australia Ltd	ans	798'/	864	Ordinary
5-May-14	UBS Securities Australia Ltd	ank	/83	8	Ordinary
May-14		ing c	9,149	340	Ordinary
+	USS Securities Australia Ltd	Buy	560	1/	Ordinary
7	UBS Securities Australia Ltd	Buy	2,048	577	Ordinary
1	UBS Securities Australia Ltd	Buy	3,149	346	Ordinary
┪	UBS Securities Australia Ltd	Buy	16,198	1,780	Ordinary
7	JBS Securities Australia Ltd	Buy	6,325	695	Ordinary
5-May-14	JBS Securities Australia Ltd	Buy	1,101	121	Ordinary
7	UBS Securities Australia Ltd	Buy	105,193	11,547	Ordinary
_	JBS Securities Australia Ltd	Sell	1,554	(169)	Ordinary
┪	JBS Securities Australia Ltd	Buy	248,426	26,994	Ordinary
7	UBS Securities Australia Ltd	Sell	374,854	(40,749)	Ordinary
	UBS Securities Australia Ltd	Sett	92,770	(10,039)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	23,258	(2,530).	Ordinary
	JBS Global Asset Management	Buy	163,147	17,772	Ordinary
_	JBS Global Asset Management	Buy	23,058	2,496	Ordinary
	JBS Global Asset Management	Buy	686'96	10,523	Ordinary
Г	JBS Global Asset Management	Buy	47,712	5.187	Ordinary
16-May-14	LIRS Global Asset Management	Bux	11 355	1 731	Ordinan

70.00		2	344	ax or	Ordinary
9-May-14	URS Securities Australia Ltd	200	33 000	2743	in in in
9-Mav-14	UBS Securities Australia I to	Riv	136	2,,72	Ordinary Ordinary
9-May 14	I BC Counties Australia 144	(c) d	207. 61	2 .	O GILINA
ti -keint-C	ODS Securities Australia Etd	, and	13,780	525,1	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	7,557	836	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	7,955	880	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	3,734	413	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	24,516	2,712	Ordinary
9-May-14	UBS Securities Australia Ltd	Sell	3,851	(426)	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,350	149	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	653	77	Ordinary
9-May-14	LIBS Securities Australia Itd	Bir	2 700	2000	Ordinary
19-May 14	1 DC Consistent Auction 144		000	25.5	Ciditialy
1	COS Commission A manufactured and	, and	060	`\	Ordinary
9-ividy- 14	Opp Securities Australia Lid	puy	5,584	/19	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	453	50	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,070	529	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	425	47	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	208	23	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	2,142	757	Ordinary
19-Mav-14	UBS Securities Australia Ltd	Birv	1 130	125	Ordina
9-May-14	I IBS Securities Australia I to	218	002	23	in in
19-May-14	1 IRC Sociution Australia 14d	600	4 196	G Ç	Ordinaly
100	COO Seconties Australia Liu	, and	4,100	403	Ordinary
9-IVIdy-14	UBS Securities Australia Ltd	Buy	2,142	237	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	262	29	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,012	112	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	22	æ	Ordinary
9-May-14	UBS Securities Australia Ltd	àc	2.088	731	Ordinary
9-May-14	I IBS Securities Australia Ltd	, and	17.0		
O May 14	1 IbC Countition Australia 1+0		2/1	h i	Countries
Plylay-14	Obs securities Australia Ltd	ons.	040,	415	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,130	125	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	452	20	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2.504	277	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	1 492	165	Ordinary
O MANY 5.4	I IDC Code services Assertation 1-4	60	201,1	200	, i
- Iviay	Coo Securities Australia Liu	hng	517'7	242	Ordinary
y-May-14	UBS Securities Australia Ltd	Buy	4,167	461	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,012	112	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	525	52	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	1.130	175	Ordinary
9-May-14	I BS Societities Australia Ltd	Bin	199		
0 142. 14	100 Commission Assessed 100 Commission Assessed		100	5	Cidillaly
9-Ividy-14	Obs Securities Australia Ltd	Buy	305	001	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	5,011	554	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	6,015	999	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	1.898	210	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	928	106	Ordinan
O May 14	LIBC Contractor Australia	650	000	3 2	ordinal y
+1-kin-C	יישים הביתווופה אתפוומוס בית	bay	88/	G	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	3,643	403	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	5,035	257	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	226	72	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	5.261	582	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buy	3 155	340	Ordinary
9-May-14	I IRS Sequenties Australia 11d	8.5	100		orien or
Q-May-14	Libe Contractor Australia 144	600	550	6	O Gillary
- Ividy	COS Securities Australia Ltd	eny	707	67	Ordinary
9-May-14	Ubs securities Australia Ltd	Buy	D51	72	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	4,339	480	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,224	246	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	3,878	429	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	868	96	Ordinary
9-May-14	UBS Securities Australia I to	Biv	2 002	232	Ordinan
D-14 200 1.4	I IDO Contration Australia I+a	5.00	100,2	207	or and
ti-kain-	ODS Securities Australia Liti	and a	184	CC.	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	127	14	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	461	51	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	172	19	Ordinary
	UBS Securities Australia Ltd	Buy	3 101	3/13	Ordinany
•	Securities	Buy	0200	355	or display
	מפרים ותפס שמפוניתום דירם	louy	10000	1 111 1	L M I M LOUI V
19-May-14					
	UBS Securities Australia Ltd	Buy	36	4	Ordinary

6-Mav-14	185 Securities Australia Ltd	la	1 388	150	Ordinan
16-May-14	UBS Securities Australia Ltd	- Sell	30,821	(3,332)	Ordinary
16-May-14	UBS Securities Australia Ltd	Sell	3,006	(325)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	9,287	(1,004)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	1,859	(201)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	814	(88)	Ordinary
16-May-14	UBS Securities Australia Ltd	Sell	37	(4)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	2,007	(217)	Ordinary
y-14	UBS Securities Australia Ltd	Sell	1,230	(133)	Ordinary
16-May-14	UBS Securities Australia Ltd	Sell	777	(84)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	92,550	(10,000)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	926	(100)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	2,409	(801)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	2,361	(522)	Ordinary
6-May-14	UBS Securities Australia Ltd	les	3,239	(350)	Ordinary
y-14	UBS Securities Australia Ltd	Sell	1,628	(176)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	2,701	(292)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	2,581	(279)	Ordinary
6-May-14	UBS Securities Australia Ltd	Self	2,239	(242)	Ordinary
6-May-14	UBS Securities Australia Ltd	Self	4.264	(461)	Ordinary
6-May-14	UBS Securities Australia Ltd	Self	1,859	(201)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	3,793	(410)	Ordinary
6-May-14	UBS Securities Australia Ltd	lles	2.812	(304)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	2.313	(250)	Ordinary
6-May-14	UBS Securities Australia Ltd	ll as	1.055	(114)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	1.758	(190)	Ordinary
y-14	UBS Securities Australia Ltd	ll es	3.219	(348)	Ordinary
6-May-14	UBS Securities Australia Ltd	3,4	1.656	(179)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	1,184	(128)	Ordinary
6-May-14	UBS Securities Australia Ltd	Self	3,127	(338)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	2,840	(302)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	956	(100)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	407	(44)	Ordinary
av-14	UBS Securities Australia Ltd	Sell	1.878	(203)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	2,738	(296)	Ordinary
16-May-14	UBS Securities Australia Ltd	Buy	181	2	Ordinary
6-May-14	UBS Securities Australia Ltd	Seil	2.800	(307)	Ordinary
-V-14	LIBS Securities Australia Ltd	i d	73	(8)	Ordinany
19-May-14	UBS Securities Australia Ltd	Sell	1.790	(197)	Ordinary
34-14	UBS Securities Australia Ltd	Buv	72,923	8,030	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	15.696	1,730	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Sell	142.305	(15,661)	Ordinary
9-May-14	I IRS Securities Australia I td	2 2	351.81	2,000	or dina
O.May.14	1 IBC Countition Australia 1+4	100	10,150	2,000	Columny
	LIDS Securities Australia Ltd	inac	10,155	(7,723)	Orginary
19-May-14	UBS Securities Australia Ltd	Enter Into Derivative d	N/A	194	Options
3y-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	200	Options
9-May-14	UBS Securities Australia Ltd	Buy	14,609	1,616	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	57,332	6,342	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	18,984	2,100	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,893	320	Ordinary
3y-14	UBS Securities Australia Ltd	Buy	325	36	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	16,941	1,874	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	83,692	9,258	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	68,360	7,562	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	363.697	40.232	Ordinary
9-May-14	1 BS Securities Australia Ltd	Buz	40 327	4 461	Ordinan
1,1	1 DC Counting Australia 1+0	600	37.480	4 1 4 6	diam'r.
19-14/dy-14	1 105 Constitutes Australia Ltd	âng	37,480	4,146	Ordinary
± .	Obs securities Australia Ltd	euy	3,553	393	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	13,560	1,500	Ordinary
9-May-14		Buy	9,130	1,010	Ordinary
y-14	UBS Securities Australia Ltd	Buy	344	38	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	12,656	1,400	Ordinary
1 4	UBS Securities Australia Ltd	Buv	5.496	808	Ordinary
9-May-14	I IRS Securities Australia I to	Big	557 001	11 143	Crainary
:	ODS Securities Australia Ltd	fing	100,735	-,145	Certain
· ·					
2-14rdy-14	UBS Securities Australia Ltd	Buy	69,961	7,739	Ordinary

9-May-14	C85 Securities Australia (10	25	2 1 1 7	750	Ordinary
	Lab Charles A section of C CC		7117	100	
19-May-14	I IRS Societities Australia Ltd	Buy	1999	1,055	Ordinary
19-M-W-14	I IRC Countries A setralia 144	Berr	000 01	000	Clumary
10 1424 14	I DC Cocurition Australia Ltd	ona)	050,61	2,000	Orginary
O March	I IDS Cocurition Australia Ltd	ouy out	000,1	/=	Orginary
10 140	Old Securities Australia Lite	And	14,909	000,1	Ordinary
12-INIdy-14	Upo Securities Australia Liu	Buy	005,081	20,000	Ordinary
10 May 14	ODS Securities Australia Ltd	end)	850/1		Ordinary
19-May 14	UBS Securities Australia Ltd	Áng	850,1 900,001	/	Grainary
S-ividy-14	Obs securities Australia Lid	Buy	008,081	70,00	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,117	234	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	226,125	25,000	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	9,045	1,000	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	9,045	1,000	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	15,141	1,674	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	9,913	1,096	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	4,523	200	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	15,811	1,748	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	905	100	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	9,045	1,000	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,131	125	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	5.653	625	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	14.438	1 598	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	1.391	154	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	4.619	511	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,085	120	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	σ	-	Ordinan
19-May-14	I BS Securities Australia I to	Bliv	5 225	873	Ordinary
O May 14	1 DC Consistor Australia Ltd	600	100	37.0	Oldinary
O May 14	1 OC Committee Australia 144	, and	08-	17	Ordinary
2-ividy-14	Obsisted Australia Ltd	, and	13,014	000	Ordinary
9-IVIAY-14	UBS Securities Australia Ltd	Buy	1,211	134	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,121	124	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,591	176	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	127	14	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,594	287	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	5,650	625	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	5,650	625	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	5.650	625	Ordinary
19-May-14	URS Securities Australia I td	Riv	020.6	220	Croiner
19-May-14	I IRS Sequenties Australia 14d	Berry	1 844	22.00	olonialy ordinary
9-May-14	I IRC Countries Ametralia 14d	600	906	5	Ordinary
10 May 14	I los con mistos Australia Lea	, and	002	57	Croinary
19-IVIAy-14	UBS Securities Australia Ltd	BOY	4,032	446	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	5,650	625	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,748	304	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	12,358	1,367	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	21,259	2,353	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	8,132	006	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	9,035	1,000	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	6.473	716	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	687	92	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	5.650	675	Ordinary
9-May-14	UBS Securities Australia Itd	Biv	σ	-	Ordinan
19-May-14	I IRC Counties Australia 14d	Brix	200	- 08	Ordinary
Q May 14	TIPE Considers Australia India	600	136.9	COL	Cidillary
10 May 14	tipe committee Australia Ltd	And S	0,333	50/	Ordinary
10 May 14	ODS Securities Australia Liu	(na	10,143	00/1	Country
3-Mdy-14	Obsisecurities Australia Ltd	Buy	1/5,1	1/4	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	797	29	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	589	32	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	15,965	1,766	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,278	252	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,130	125	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	996	107	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	8,136	006	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	6,183	684	Ordinary
9-May-14					
	I IRS Securities Australia Ltd	~ 2	1 255	130	Ordinary

1 1 1	UBS Securities Australia Ltd	Buy	113,823	12,591	Ordinary
- 1					
	UBS Securities Australia Ltd	Buy	1,012	112	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	905	100	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	296	107	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,085	120	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	461	51	Ordinary
May-14	UBS Securities Australia Ltd	Buy	72,360	8,000	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	12,168	1,346	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	913	101	Ordinary
- 1	UBS Securities Australia Ltd	Buy	1,284	142	Ordinary
	UBS Securities Australia Ltd	Buy	1,618	179	Ordinary
	UBS Securities Australia Ltd	Buy	8,733	996	Ordinary
t	UBS Securities Australia Ltd	Buy	17,547	1,941	Ordinary
ı	UBS Securities Australia Ltd	Buy	21,624	2.392	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Ban	15,006	1 660	Ordinary
ı	UBS Securities Australia I td	Biv	11 680	1 292	Ordinan
1	I BS Securities Australia I to	Biz	1 781	107	Ording
19.6434.14	LISC Countries Australia Ltd	600	032	20	Ordinary
	I IBC Cocurities Australia 14d	600	007	314	Ordinary
0 May 14	100 Cocurition Australia 14d	B. iv	307/5	2 5	Oldinary Oldinary
0 May 14	1 IDC Commission American Lea	, in a	183	601	Ordinary
O May 14	I DC Countries Australia Ltd	200	1 2 2	117	Ordinary
	The second secon	900	/co.,		O'CILIBILY
May 14	UBS Securities Australia Ltd	eny	1,102	122	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	7,535	834	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	21,533	2,382	Ordinary
- 1	UBS Securities Australia Ltd	Buy	841	93	Ordinary
	UBS Securities Australia Ltd	Buy	1,582	175	Ordinary
	UBS Securities Australia Ltd	Buy	9,411	1,041	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,658	294	Ordinary
1	UBS Securities Australia Ltd	Buy	3.453	382	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buv	25.366	2,806	Ordinary
19-May-14	UBS Securities Australia Ltd	Biv	22 555	2 495	Ordinan
1	I IBS Securities Australia I to	Bix	1 546	171	Ordinary
	I IRC Securities Australia 14d	(5)	120 00	0030	Ordinary Ordinary
19-14:dy-14	Upo Securities Australia Lid	Áng	700,7	6707	Ordinary
- 1	Obs Securities Australia Ltd	áng	5,102	coc	Ordinary
9-May-14	UBS Securities Australia Ltd	gny	13,2/4	1,4/0	Ordinary
- 1	UBS Securities Australia Ltd	Buy	4,330	479	Ordinary
	UBS Securities Australia Ltd	Buy	5,930	929	Ordinary
	UBS Securities Australia Ltd	Buy	904	100	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	1,058	117	Ordinary
ı	UBS Securities Australia Ltd	Buv	6.256	269	Ordinary
ı	I IBC Consisting Australia Ltd	0,00	C	-	Ordinany
May-14	Obs securities Australia Ltd	ing.	6	-	Ordinary
- 1	UBS Securities Australia Ltd	Buy	29,507	3,264	Ordinary
- 1	UBS Securities Australia Ltd	Buy	823	91	Ordinary
	UBS Securities Australia Ltd	Buy	6,120	229	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	136	15	Ordinary
ŀ	UBS Securities Australia Ltd	Buy	4.303	476	Ordinary
14 veh	I IBS Societies Australia I to	Bir	F 261	582	Ording
10 110	I DC Comition Australia Ltd	600	102,0	200	() () () () () ()
- 1	Obs securities Australia Liu	hod	4,662	040	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	5,261	285	Ordinary
	UBS Securities Australia Ltd	Buy	2,974	329	Ordinary
ı.	UBS Securities Australia Ltd	Buy	108	12	Ordinary
	UBS Securities Australia I td	Buy	1 003	111	Ordinary
10-May-14	I IRC Congression American 14d	(-10	2006	376	Ordinan
vidy- i -	OBS SECULIES AUSUAIIS EIG	ing	2,400	6/7	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	SOR'E	437	Ordinary
May-14	UBS Securities Australia Ltd	Buy	89,618	806'6	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	3,987	441	Ordinary
19-Mav-14	UBS Securities Australia Ltd	Suv	82	2	Ordinary
9-Mav-14	I IBS Securities Australia I td	Biv	7.874	871	Ordinary
	190 Commission Assessed and		300	Š	(Files
9-May-14	UBS Securities Australia Ltd	àng	977	57	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	407	45	Ordmary
9-May-14	UBS Securities Australia Ltd	Buy	1,817	201	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	3,155	349	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	8,570	948	Ordinary
19-May-14	De Constitue Australia (+)				
Vidy-14	The state of the s	•	14 050	1 655	Creditor

Prividy-14	1 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				
9-May-14	LIRS Securities Australia 14d	Buy	710	35	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	7.376	2 12	Ordinary
19-May-14	UBS Securities Australia Ltd	Birk	0,5,7	137	Ordinary
19-May-14	I BS Securities Australia I to	i i	1 020	è	Guillary
19-May-14	LIBS Securities Australia Ltd	Buck	1,020	2002	Ordinary
19-Mav-14	UBS Securities Australia I to	Bary	1 719	190	Colone
19-Mav-14	UBS Securities Australia I1d	Bity	190	1,0	Continue
19-May-14	LIBS Securities Australia I to	Bin	900	110	Ordinary
19-May-14	UBS Securities Australia I td	Bin	1881	2000	Ordinay.
19-May-14	I IRS Securities Australia 14d	2 2	1000	210	Ordinary
19-May-14	I IRS Societies Australia 11d	60	202	212	Couriery
19-May-14	LIBC Counties Australia 14d	900	2000	ò	Ordinary
10 May 14	Tibe committee Australia Ltd	Second	795,1	151	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,520	168	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	22,851	2,525	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,257	139	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	3,330	368	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,425	268	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	4,579	909	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,937	214	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	13,947	1,542	Ordinary
1	UBS Securities Australia Ltd	Buy	1,901	210	Ordinary
	UBS Securities Australia Ltd	Buy	1,267	140	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	49,856	5,512	Ordinary
	UBS Securities Australia Ltd	Buy	2,281	252	Ordinary
	UBS Securities Australia Ltd	Buy	2,977	329	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	272	30	Ordinary
19-May-14	UBS Securities Australia Ltd	Box	1.493	165	Ordinary
1	UBS Securities Australia Ltd	Buy	1 710	180	Ordinary
	UBS Securities Australia Ltd	Buy	099	73	Ordinary
19-Mav-14	UBS Securities Australia Ltd	B. iv	37	^	Ordinan
1	UBS Securities Australia Ltd	Biry	895	9	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	692	88	Ordinary
19-Mav-14	UBS Securities Australia 11d	Bis	163	100	ording.
19-May-14	LIBS Securities Australia Ltd	200	1 557	17.	Goldinary
i	LIBS Securities Australia Ltd	Bis	ō	-	Ordinary
19-May-14	I BS Securities Australia I to	Bitt	2 263	. 020	ording.
	Let Contract August 1-10	600	20272	250	Guraly
1	1 IDC Countries Australia Ltd	Son S	467,1	900	Couriery
10 May 14	Libe Contribute Australia Lib	buy B	447 447	772	Ordinary
10 Mar. 14	Obs securities Australia Ltd	hod	3,394	3/5	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,367	151	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	905	100	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,842	314	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	905	100	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,104	122	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	5,023	222	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,534	280	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	1,005	11	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	1.086	120	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	1.086	120	Ordinary
19-May-14	UBS Securities Australia Ltd	Bliv	10.036	1 109	Ordinary
19-Mav-14	UBS Securities Australia I td	Buy	10 534	1 164	Ordinary
19-May-14	UBS Securities Australia I td	Bliv	1 367	151	Ordinary
19-May-14	LIBS Sequities Australia Ltd	Bark	327 07	5 467	Ordinary
19-May-14	IBS Consisting Augustia 14d	600	ACT AC	000 0	Oldinary
19-May-14	IIR Sequenties Australia 14	and a	04,74C	2,030	Ordings
10 6401 14	The Contraction Australia Lea	pa)	064'ı	130	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	21,254	2,383	Ordings
O-May 14	ılن	500	6 10	101	Oldinary
19-May-14	IIBC Coourties Australia Ind	poo.	914	107.0	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	667'57	7,797	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	8,150	006	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	543	09	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	915	101	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,331	147	Ordinary
9-Mav-14	Of Contraction Assessed 140				The second secon
	ODD SECULIES AUStralia LIU	Buy	18,029	1,990	Ordinary

	CDD SECTIONS AND INC.	2			Ordinary
19-May-14	1 IDC Commission Assessed and	653	0.0		
9-May-14	UBS Securities Australia Ltd	Buy Bus	5,840		Ordinary
10-May 14	198 Societies Australia 14d	bug.	000'1		Oranary
-Mdy-14	UBS Securities Australia Ltd	Buy	9,040		Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	24,516	2,	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	4,701	520	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	6,120	677	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	29,714	3,287	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	108	12	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	1,427	158	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buv	12 498		Ordinary
9-May-14	I IBS Securities Australia I tol	Bliv	020 20	100.0	Credinary
1	Data Street, A contract of the	600	0/6/12		Organs
tolon-	Opp Securities Australia Ltu	λη _α	651,13	3,000	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	5,499	909	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,107	233	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	9,045	1,000	Ordinary
9-May-14	UBS Securities Australia Ltd	BC	706	78	Ordinary
9-Mav-14	LIBS Securities Australia Ltd	Rrv	25 353	2 803	Ordinary
O Mari 14	LIDS Consisting Australia (+A	, , ,	134 64	200,2	O'GIII G
1	Library A - 11-14-1	600	104/64		Comary
3-May-14	Ubs securities Australia Ltd	Buy	850,1		Ordinary
y-May-14	UBS Securities Australia Ltd	Buy	31,730		Ordinary
9-May-14	IUBS Securities Australia Ltd	Buy	31,269	3,457	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,098	232	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	9.045	0001	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buv	8 141	UUG	Ordinary
O May 14	I IBC Countries Australia I tol	Do an	ACC A	237	Ordinos.
O May 14	TIDE Countries Australia 14d	D. T.	1777	101	Oldinary
-Ividy 14	ODD Deculties Australia Liu	and C	2,10/	523	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,107		Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	5,210		Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	13,531	1,496	Ordinary
9-Mav-14	UBS Securities Australia Itd	Biv	1 284		Ordinary
0 h/m/ 1.4	I IDC Committee Australia I tol	Bine	6 744		
	CO COCCUERCO POST CALL	ing.	111/0		Cluiriary
9-May-14	UBS Securities Australia Ltd	Buy	11,915	1,318	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	5,867	649	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	1,799	661	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	22,600	2,500	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buy	10.206	1179	Ordinary
Q-May-14	I IBS Securities Australia I to	Buy	000	373	Ordings
Thidy-in-	Upp occupies Australia Ltd	and broke	0,0,0	000	Granary
9-May-14	UBS Securities Australia Ltd	Buy	5,840	646	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	29,733	3,289	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	22,600	2,500	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	18.522	2.050	Ordinary
9-May-14	LIBS Securities Australia Ltd	BIR	1130	125	Ordinany
1	COO Securities Australia Liu	and a	OE!'	671	Cidirlary
19-May-14	UBS Securities Australia Ltd	Buy	443	49	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	10,794	1,194	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	26.098	2.887	Ordinary
May. 14	I IBC Constitute Australia 14d	D. W.	10000	000 0	Ordinary)
9-Wdy-14	ODS Securities Australia LIG	euy	060,81	7,000	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,031	114	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	904	100	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	16,525	1,829	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,419	157	Ordinary
9-May-14	I IBS Sequition Australia I to	8.17	50 271	5 561	Ordinany
O 6400 14	I IDC Coontistor Australia Ltd		4 000	.00%	Cicil di
-loldy-14	ODS Seculities Australia Ltd	ina c	nn,4	222	Oranary
9-May-14	UBS Securities Australia Ltd	Buy	4,041	44/	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	27,192	3,008	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	052	83	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	5.966	099	Ordinary
Q-1434-14	I IBC Copurities Australia 14d	D. D.	20 505	036.6	Ordinary
9-May-14	Upo Securities Australia Lid	ána.	5050	897'7	Ordinary
-May-14	UBS Securities Australia Ltd	any	806/6	960'1	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	9,139	1,011	Ordinary
-May-14	UBS Securities Australia Ltd	Buy	5,749	969	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	19,620	2,168	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	3,892	430	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buy	1.032	114	Ordinary
19-Mav-14	UBS Securities Australia I td	Buy	127	14	Ordinary
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	The first Australia Etc	, and	/21	-	Commany
19-May-14	UBS Securities Australia Ltd	Buy	6	-	Challoary
			_		()

OS Securities Autorials LIVE By 5,1231 CORREST DES Securities Autorials LIVE By 6,100 5,1241 CORREST DES Securities Autorials LIVE By 6,100 6,100 4,771 DES Securities Autorials LIVE By 6,100 6,100 6,171 DES Securities Autorials LIVE By 1,100 1,172 1,101 DES Securities Autorials LIVE By 1,102 1,107 1,107 DES Securities Autorials LIVE By 1,102 1,107 <th>0 May 14</th> <th>I IDC Commission A metrolia 1 to</th> <th></th> <th></th> <th></th> <th>200</th>	0 May 14	I IDC Commission A metrolia 1 to				200
USS Securities Australia Ltd	3-May-14	I IRC Securities Australia Ltd	pan)	51 804	757 3	Cicinion
By 618 By 1,12 By	9-May-14	UBS Securities Australia Ltd	Buy	43,256	4.777	Ordinary
(BS Securities Autorials LTD (By (1,558) (11) (BS Securities Autorials LTD (By (1,528) (1,521) (1,521) (BS Securities Autorials LTD (By (1,524) (1,521) (1,521) (1,521) (BS Securities Autorials LTD (By (By <td>3-May-14</td> <td>UBS Securities Australia Ltd</td> <td>) Ag</td> <td>616</td> <td>89</td> <td>Ordinary</td>	3-May-14	UBS Securities Australia Ltd) Ag	616	89	Ordinary
Bay 1,221 1,221 1,221 1,221 1,221 1,221 1,221 1,222 1,222 1,223	3-May-14	UBS Securities Australia Ltd	Buy	1.059	117	Ordinary
Box Securities Australia Ltd Buy 1,521 1,501 1,501 1,502	3-May-14	UBS Securities Australia Ltd	Buy	8,326	919	Ordinary
Reservoires Autoriale LEG Bay 1,622 2,555 LES Securities Autoriale LEG Buy 1,624 1,625 LES Securities Autoriale LEG Buy 1,624 1,627 LES Securities Autoriale LEG Buy 1,627 1,627 LES Securities Autoriale LEG Buy 1,625 1,627 LES Securities Autoriale LEG Buy 6,620 1,627 LES Securities Autoriale LEG Buy 6,620 1,627 LES Securities Autoriale LEG Buy 1,625 1,627 LES Securities Autoriale LEG Buy 1,627 2,627 LES Securities Autoriale LEG	3-May-14	UBS Securities Australia Ltd	Buv	172	- 61	Ordinary
(BSS Securities Australia Ltd BLY 185 F88 2.055 (BSS Securities Australia Ltd BLY 1.640 1.82 (BSS Securities Australia Ltd BLY 1.640 1.82 (BSS Securities Australia Ltd BLY 1.671 1.02 (BSS Securities Australia Ltd BLY 1.572 1.02 (BSS Securities Australia Ltd BLY 1.522 1.68 (BSS Securities Australia Ltd BLY 2.500 2.50 (BSS Securities Australia Ltd BLY 2.500 2.50 (BSS Securities Australia Ltd BLY 2.500 2.50 (BSS Securities Australia Ltd BLY 2.500 2.73 (BSS Securities Australia Ltd BLY 2.500 2.73 (BSS Securities Australia Ltd BLY 2.500 2.78 (BSS Securities Australia Ltd	9-May-14	UBS Securities Australia Ltd	Buv	1821	201	Ordinary
(BSS Securities Australia tot Buy 3,666 198 (BSS Securities Australia tot Buy 1,151 1,27 (BSS Securities Australia tot Buy 1,151 1,27 (BSS Securities Australia tot Buy 5,200 245 (BSS Securities Australia tot Buy 6,601 1,527 (BSS Securities Australia tot Buy 6,603 1,527 (BSS Securities Australia tot Buy 1,627 1,527 (BSS Securities Australia tot Buy 1,627 1,527 (BSS Securities Australia tot Buy 1,550 1,500 (BSS Securities Australia tot Buy 2,520 2,83 (BSS Securities Australia tot Buy 2,500 1,500 (BSS Securities Australia tot Buy 2,500 1,000 (BSS Securities Australia tot Buy 2,500 2,000 (BSS Securities Australia tot Buy 2,500 2,000 (BSS Securities Australia tot Buy 2,500 2,000 (BSS Securities Australia tot	9-May-14	UBS Securities Australia Ltd	Buv	18.618	2.055	Ordinary
(BSS Securities Australia Ltd Buy 1,559 182 (BSS Securities Australia Ltd Buy 1,571 1,771 (BSS Securities Australia Ltd Buy 6,016 6,64 (BSS Securities Australia Ltd Buy 2,220 1,523 (BSS Securities Australia Ltd Buy 2,220 1,523 (BSS Securities Australia Ltd Buy 2,220 1,523 (BSS Securities Australia Ltd Buy 1,522 1,524 (BSS Securities Australia Ltd Buy 1,542 1,527 (BSS Securities Australia Ltd Buy 1,542 1,527 (BSS Securities Australia Ltd Buy 1,542 1,507 (BSS Securities Australia Ltd Buy 1,542 1,507 (BSS Securities Australia Ltd Buy 1,542 1,507 (BSS Securities Australia Ltd Buy 2,508 2,788 (BSS Securities Australia Ltd Buy 1,500 1,507 (BSS Securities Australia Ltd Buy 2,508 2,788 (BSS Securities Australia Ltd </td <td>3-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>3.606</td> <td>398</td> <td>Ordinary</td>	3-May-14	UBS Securities Australia Ltd	Buv	3.606	398	Ordinary
(BSS Securities Australia Ind Buy 9.24 10.2 (BSS Securities Australia Ind Buy 1.5.30 1.6.7 (BSS Securities Australia Ind Buy 6.016 6.64 (BSS Securities Australia Ind Buy 6.017 7.2.30 1.2.73 (BSS Securities Australia Ind Buy 1.0.455 1.1.524 1.0.72 (BSS Securities Australia Ind Buy 1.0.456 1.0.74 1.0.74 (BSS Securities Australia Ind Buy 1.0.74 1.0.74 1.0.74 (BSS Securities Australia Ind Buy 1.0.74 1.0.74 1.0.74 (BSS Securities Australia Ind Buy 2.7.70 2.8.37 1.0.74 (BSS Securities Australia Ind Buy 2.7.70 2.8.37 1.0.70 (BSS Securities Australia Ind	3-May-14	UBS Securities Australia Ltd	Buv	1,649	182	Ordinary
(BSS Securities Australia Ltd BLY (15.51 12.7 (BSS Securities Australia Ltd Buy (2.200 664 (BSS Securities Australia Ltd Buy (2.200 24.53 (BSS Securities Australia Ltd Buy (2.200 26.25 (BSS Securities Australia Ltd Buy (1.542) 1.624 (BSS Securities Australia Ltd Buy (1.542) 1.627 (BSS Securities Australia Ltd Buy (2.510) 2.78 2.78 (BSS Securities Australia Ltd Buy (2.510) 2.70 2.78 (BSS Securities Australia Ltd Buy (2.510) 2.78 2.78 (BSS Securities Australia Ltd Buy (2.510) </td <td>3-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Six</td> <td>426</td> <td>107</td> <td>Ordinary</td>	3-May-14	UBS Securities Australia Ltd	Six	426	107	Ordinary
UBS Securities Australia Ltd Bby 6,016 6,62 UBS Securities Australia Ltd Bby 1,223 1,273 UBS Securities Australia Ltd Bby 1,625 1,54 UBS Securities Australia Ltd Bby 1,625 1,65 UBS Securities Australia Ltd Bby 1,624 1,574 UBS Securities Australia Ltd Bby 1,624 1,574 UBS Securities Australia Ltd Bby 1,642 1,574 UBS Securities Australia Ltd Bby 1,4324 1,627 UBS Securities Australia Ltd Bby 1,4324 1,657 UBS Securities Australia Ltd Bby 1,540 1,10 UBS Securities Australia Ltd Bby 2,508 2,518 UBS Securities Australia Ltd Bby 2,508 1,65 UBS Securities Australia Ltd Bby 1,500 1,70 UBS Securities Australia Ltd Bby 1,500 1,70 UBS Securities Australia Ltd Bby 1,500 1,70 UBS Securities Australia Ltd Bby	3-May-14	UBS Securities Australia Ltd	Buv	1.151	127	Ordinary
UBS Securities Australia Ltd Bby 1,522 2,520 2,52 UBS Securities Australia Ltd Buy 1,523 1,52 UBS Securities Australia Ltd Buy 1,62 1,52 UBS Securities Australia Ltd Buy 1,52 1,52 UBS Securities Australia Ltd Buy 5,66 6.23 UBS Securities Australia Ltd Buy 1,57 3,73 UBS Securities Australia Ltd Buy 2,50 2,73 UBS Securities Australia Ltd Buy 4,39 4,39 UBS Securities Australia Ltd Buy 4,39 4,39 UBS Securities Australia Ltd Buy 4,59 2,78 UBS Securities Australia Ltd Buy 4,59 2,78 1,18 UBS Securities Australia Ltd Buy 1,57 1,18 UBS Securities Australia L	9-May-14	UBS Securities Australia Ltd	Buv	6.016	664	Ordinary
UBIS Securities Australia Ltd Bay 1,273 1,273 Bob 1,524 1,542 1,542 UBS Securities Australia Ltd Bay 1,542 1,542 UBS Securities Australia Ltd Bay 1,542 1,543 UBS Securities Australia Ltd Bay 1,544 1,527 UBS Securities Australia Ltd Bay 1,544 1,527 UBS Securities Australia Ltd Bay 2,573 3,588 UBS Securities Australia Ltd Bay 2,570 3,588 UBS Securities Australia Ltd Bay 3,520 3,68 UBS Securities Australia Ltd Bay 3,520 3,68 UBS Securities Australia Ltd Bay 3,524 1,67 UBS Securities Australia Ltd Bay 3,520 3,68 UBS Securities Australia Ltd Bay 3,	3-May-14	UBS Securities Australia Ltd	and Buy	2,220	245	Ordinary
Biss Securities Australia Ltd Bay 1,525 1,528 Biss Securities Australia Ltd Buy 5,683 1,154 Biss Securities Avaitaria Ltd Buy 5,683 1,274 Biss Securities Avaitaria Ltd Buy 2,483 1,570 Biss Securities Avaitaria Ltd Buy 2,483 1,570 Biss Securities Avaitaria Ltd Buy 6,734 1,570 Biss Securities Avaitaria Ltd Buy 6,734 1,570 Biss Securities Avaitaria Ltd Buy 6,734 1,514 Biss Securities Avaitaria Ltd Buy 4,520 1,67 Biss Securities Avaitaria Ltd Buy 4,520 2,18 Biss Securities Avaitaria Ltd Buy 4,520 1,67 Biss Securities Avaitaria Ltd Buy 1,520 1,67 Biss Securities Avaitaria Ltd Buy 1,530 1,68 Biss Securities Avaitaria Ltd Buy 1,520 1,68 Biss Securities Avaitaria Ltd Buy 1,530 1,68 Biss Securities Avaitaria Ltd	3-May-14	UBS Securities Australia Ltd	Buv	11.533	1.273	Ordinary
UBIS Securities Australia Ltd Bby 1,522 168 UBIS Securities Australia Ltd Buy 1,623 1,574 UBIS Securities Australia Ltd Buy 1,644 1,274 UBIS Securities Australia Ltd Buy 1,4744 1,570 UBIS Securities Australia Ltd Buy 29,734 3,788 UBIS Securities Australia Ltd Buy 29,708 3,885 UBIS Securities Australia Ltd Buy 29,708 2,837 UBIS Securities Australia Ltd Buy 29,708 3,885 UBIS Securities Australia Ltd Buy 2,570 3,88 UBIS Securities Australia Ltd Buy 1,540 1,70 UBIS Securities Australia Ltd Buy 2,570 1,667 UBIS Securities Australia Ltd Buy 1,524 1,687 UBIS Securities Australia Ltd Buy 1,524 1,10 UBIS Securities Australia Ltd Buy 1,524 1,10 UBIS Securities Australia Ltd Buy 1,524 1,10 UBIS Securities Australia Ltd<	3-May-14	UBS Securities Australia Ltd	Buv	10,455	1.154	Ordinary
UBS Securities Australia Ltd Buy 5.63 672 UBS Securities Australia Ltd Buy 1,1542 1,274 UBS Securities Australia Ltd Buy 24,833 2,741 UBS Securities Australia Ltd Buy 1,3590 1,500 UBS Securities Australia Ltd Buy 67,343 1,273 UBS Securities Australia Ltd Buy 4,394 1,510 UBS Securities Australia Ltd Buy 2,510 2,73 UBS Securities Australia Ltd Buy 4,530 1,70 UBS Securities Australia Ltd Buy 4,530 1,70 UBS Securities Australia Ltd Buy 4,530 1,70 UBS Securities Australia Ltd Buy 1,517 1,11 UBS Securities Australia Ltd Buy 1,520 1,08 UBS Securities Australia Ltd Buy 1,520 1,08 UBS Securities Australia Ltd Buy 1,520 1,09 UBS Securities Australia Ltd Buy 1,520 1,08 UBS Securities Australia Ltd Buy </td <td>3-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>1,522</td> <td>168</td> <td>Ordinary</td>	3-May-14	UBS Securities Australia Ltd	Buy	1,522	168	Ordinary
UBS Securities Australia Ltd Buy 1,542 1,274 UBS Securities Australia Ltd Buy 2,4,833 2,741 UBS Securities Australia Ltd Buy 1,542 1,570 UBS Securities Australia Ltd Buy 1,7,424 1,6272 UBS Securities Australia Ltd Buy 2,7,689 2,885 UBS Securities Australia Ltd Buy 2,500 2,877 UBS Securities Australia Ltd Buy 2,500 2,877 UBS Securities Australia Ltd Buy 2,500 2,878 UBS Securities Australia Ltd Buy 2,500 1,70 UBS Securities Australia Ltd Buy 2,500 1,70 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,524 1,667 UBS Securities Australia Ltd Buy 1,524 1,18 UBS Securities Australia Ltd Buy 1,524 1,45 UBS Securities Australia Ltd Buy 1,524 1,45 UBS Securities Australia Ltd	3-May-14	UBS Securities Australia Ltd	Buv	5.663	625	Ordinary
UBS Securities Australia Ltd Buy 2.2 48.33 2.741 UBS Securities Australia Ltd Buy 1.3,590 1,500 UBS Securities Australia Ltd Buy 67.343 7.433 UBS Securities Australia Ltd Buy 2.510 2.71 UBS Securities Australia Ltd Buy 2.510 2.71 UBS Securities Australia Ltd Buy 2.50 2.3 UBS Securities Australia Ltd Buy 2.50 2.7 UBS Securities Australia Ltd Buy 4.394 4.85 UBS Securities Australia Ltd Buy 4.394 4.85 UBS Securities Australia Ltd Buy 1.50 1.70 UBS Securities Australia Ltd Buy 1.524 1.65 UBS Securities Australia Ltd Buy <td>9-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>11,542</td> <td>1,274</td> <td>Ordinary</td>	9-May-14	UBS Securities Australia Ltd	Buy	11,542	1,274	Ordinary
UBS Securities Australia Ltd Buy 13.590 1,500 UBS Securities Australia Ltd Buy 67,342 16,272 UBS Securities Australia Ltd Buy 67,348 7,433 UBS Securities Australia Ltd Buy 25,1078 2,183 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,512 1,67 UBS Securities Australia Ltd Buy 3,137 1,118 UBS Securities Australia Ltd Buy 3,120 3,08 UBS Securities Australia Ltd Buy 1,512 1,165 UBS Securities Australia Ltd Buy 1,512 1,118 UBS Securities Australia Ltd Buy 1,512 1,165 UBS Securities Australia Ltd	9-May-14	UBS Securities Australia Ltd	Buy	24,833	2,741	Ordinary
UBS Securities Australia Ltd Buy 147,424 16,272 UBS Securities Australia Ltd Buy 67,343 7,433 UBS Securities Australia Ltd Buy 29,789 3,288 UBS Securities Australia Ltd Buy 4,394 485 UBS Securities Australia Ltd Buy 2,510 27,77 UBS Securities Australia Ltd Buy 2,520 2,633 UBS Securities Australia Ltd Buy 2,520 2,633 UBS Securities Australia Ltd Buy 2,520 2,633 UBS Securities Australia Ltd Buy 2,520 2,183 UBS Securities Australia Ltd Buy 3,153 3,48 UBS Securities Australia Ltd Buy 3,524 1,56 UBS Securities Australia Ltd Buy 3,734 1,18 UBS Securities Australia Ltd Buy 3,734 1,435 1,68 UBS Securities Australia Ltd Buy 3,724 5,76 1,18 UBS Securities Australia Ltd Buy 3,724 5,76 1,18	3-May-14	UBS Securities Australia Ltd	Buy	13,590	1,500	Ordinary
UBS Securities Australia Ltd Buy 67,243 7,433 UBS Securities Australia Ltd Buy 29,789 3,288 UBS Securities Australia Ltd Buy 4,398 3,288 UBS Securities Australia Ltd Buy 2,510 277 UBS Securities Australia Ltd Buy 2,510 20,78 UBS Securities Australia Ltd Buy 2,510 20 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 2,510 20 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,524 1,65 UBS Securities Australia Ltd Buy 1,540 1,10 UBS Securities Australia Ltd Buy 1,540 1,10 UBS Securities Australia Ltd Buy 1,540 1,10 UBS Securities Australia Ltd Buy	3-May-14	UBS Securities Australia Ltd	Buv	147.424	16,272	Ordinary
UBS Securities Australia Ltd Buy 29.789 3.288 UBS Securities Australia Ltd Buy 2.570 2.77 UBS Securities Australia Ltd Buy 2.570 2.77 UBS Securities Australia Ltd Buy 2.570 2.77 UBS Securities Australia Ltd Buy 2.570 2.78 UBS Securities Australia Ltd Buy 2.58 2.71 UBS Securities Australia Ltd Buy 2.78 2.71 UBS Securities Australia Ltd Buy 2.78 3.48 UBS Securities Australia Ltd Buy 3.153 3.48 UBS Securities Australia Ltd Buy 3.153 3.48 UBS Securities Australia Ltd Buy 3.728 4.11 UBS Securities Australia Ltd Buy 3.728 4.11 UBS Securities Australia Ltd Buy 3.728 4.11 UBS Securities Australia Ltd Buy 3.728 4.10 UBS Securities Australia Ltd Buy 3.728 4.11 UBS Securities Australia Ltd Buy	9-May-14	UBS Securities Australia Ltd	Buy	67,343	7,433	Ordinary
URS Securities Australia Ltd Buy 4,394 465 UBS Securities Australia Ltd Buy 2,510 277 UBS Securities Australia Ltd Buy 25,107 2,133 UBS Securities Australia Ltd Buy 19,778 2,133 UBS Securities Australia Ltd Buy 4,530 50 UBS Securities Australia Ltd Buy 4,530 50 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,524 1,514 UBS Securities Australia Ltd Buy 1,524 1,667 UBS Securities Australia Ltd Buy 1,524 1,667 UBS Securities Australia Ltd Buy 1,524 1,105 UBS Securities Australia Ltd Buy <td>9-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>29,789</td> <td>3,288</td> <td>Ordinary</td>	9-May-14	UBS Securities Australia Ltd	Buy	29,789	3,288	Ordinary
URS Securities Australia Ltd Buy 25,10 277 UBS Securities Australia Ltd Buy 25,10/38 2,837 UBS Securities Australia Ltd Buy 25,008 2,33 UBS Securities Australia Ltd Buy 2,50 13 UBS Securities Australia Ltd Buy 4,530 10 UBS Securities Australia Ltd Buy 1,540 1,0 UBS Securities Australia Ltd Buy 7,638 843 UBS Securities Australia Ltd Buy 1,537 1,514 UBS Securities Australia Ltd Buy 1,530 1,667 UBS Securities Australia Ltd Buy 1,512 1,667 UBS Securities Australia Ltd Buy 1,512 1,67 UBS Securities Australia Ltd Buy 1,524 1,67 UBS Securities Australia Ltd Buy 1,522 1,67 UBS Securities Australia Ltd Buy 1,535 1,67 UBS Securities Australia Ltd Buy 1,535 1,67 UBS Securities Australia Ltd Buy	9-May-14	UBS Securities Australia Ltd	Buy	4,394	485	Ordinary
UBS Securities Australia Ltd Buy 257,078 28,375 UBS Securities Australia Ltd Buy 19,778 2,183 UBS Securities Australia Ltd Buy 4,530 5.03 UBS Securities Australia Ltd Buy 4,530 5.03 UBS Securities Australia Ltd Buy 1,540 17 UBS Securities Australia Ltd Buy 7,638 848 UBS Securities Australia Ltd Buy 7,638 848 UBS Securities Australia Ltd Buy 7,638 848 UBS Securities Australia Ltd Buy 7,536 1,088 UBS Securities Australia Ltd Buy 1,524 1,65 UBS Securities Australia Ltd Buy 1,524 1,65 UBS Securities Australia Ltd Buy 1,224 1,65 UBS Securities Australia Ltd Buy 1,728 1,65 UBS Securities Australia Ltd Buy 3,728 1,78 UBS Securities Australia Ltd Buy 1,395 1,65 UBS Securities Australia Ltd Buy	9-May-14	UBS Securities Australia Ltd	Buy	2,510	277	Ordinary
URS Securities Australia Ltd Buy 19778 2,183 URS Securities Australia Ltd Buy 4,530 500 URS Securities Australia Ltd Buy 4,530 500 URS Securities Australia Ltd Buy 1,540 1,717 URS Securities Australia Ltd Buy 1,540 1,717 URS Securities Australia Ltd Buy 1,512 1,514 URS Securities Australia Ltd Buy 1,512 1,514 URS Securities Australia Ltd Buy 1,524 1,667 URS Securities Australia Ltd Buy 1,524 1,167 URS Securities Australia Ltd Buy 1,014 1,118 URS Securities Australia Ltd Buy 1,014 1,105 URS Securities Australia Ltd Buy 1,014 1,105 URS Securities Australia Ltd B	3-May-14	UBS Securities Australia Ltd	Buy	257.078	28,375	Ordinary
URS Securities Australia Ltd Buy 228 23 URS Securities Australia Ltd Buy 4,530 500 UBS Securities Australia Ltd Buy 1,540 170 UBS Securities Australia Ltd Buy 1,540 170 UBS Securities Australia Ltd Buy 7,638 843 UBS Securities Australia Ltd Buy 7,638 843 UBS Securities Australia Ltd Buy 1,571 1,514 UBS Securities Australia Ltd Buy 1,572 1,667 UBS Securities Australia Ltd Buy 1,524 1,68 UBS Securities Australia Ltd Buy 1,512 1,67 UBS Securities Australia Ltd Buy 1,734 1,18 UBS Securities Australia Ltd Buy 3,724 1,58 UBS Securities Australia Ltd Buy 1,315 1,48 UBS Securities Australia Ltd Buy 1,495 1,695 UBS Securities Australia Ltd Buy 3,724 4,28 UBS Securities Australia Ltd Buy <	9-May-14	UBS Securities Australia Ltd	Buy	19,778	2,183	Ordinary
UBS Securities Australia Ltd Buy 4,530 500 UBS Securities Australia Ltd Buy 1,540 17 UBS Securities Australia Ltd Buy 3,153 348 UBS Securities Australia Ltd Buy 7,534 848 UBS Securities Australia Ltd Buy 7,538 843 UBS Securities Australia Ltd Buy 7,548 1,687 UBS Securities Australia Ltd Buy 1,524 1,687 UBS Securities Australia Ltd Buy 3,728 1,088 UBS Securities Australia Ltd Buy 3,728 1,168 UBS Securities Australia Ltd Buy 1,524 1,65 UBS Securities Australia Ltd Buy 3,728 1,16 UBS Securities Australia Ltd Buy 1,395 1,64 UBS Securities Australia Ltd Buy 1,795 1,65 UBS Securities Australia Ltd Buy 3,724 4,28 UBS Securities Australia Ltd Buy 3,728 1,65 UBS Securities Australia Ltd Buy	3-May-14	UBS Securities Australia Ltd	Buy	208	23	Ordinary
UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,540 1,70 UBS Securities Australia Ltd Buy 1,517 1,514 UBS Securities Australia Ltd Buy 1,571 1,514 UBS Securities Australia Ltd Buy 1,524 1,667 UBS Securities Australia Ltd Buy 1,524 1,667 UBS Securities Australia Ltd Buy 1,524 1,1667 UBS Securities Australia Ltd Buy 1,524 1,116 UBS Securities Australia Ltd Buy 1,524 1,1667 UBS Securities Australia Ltd Buy 1,524 1,1667 UBS Securities Australia Ltd Buy 1,1315 1,167 UBS Securities Australia Ltd Buy 1,0140 1,118 UBS Securities Australia Ltd Buy 1,035 1,109 UBS Securities Australia Ltd Buy 1,036 1,019 UBS Securities Australia Ltd Buy 3,035 3,035 UBS Securities Australia Ltd	3-May-14	UBS Securities Australia Ltd	Buv	4.530	200	Ordinary
UBS Securities Australia Ltd Buy 1,540 170 UBS Securities Australia Ltd Buy 1,540 170 UBS Securities Australia Ltd Buy 7,638 843 UBS Securities Australia Ltd Buy 1,532 1,68 UBS Securities Australia Ltd Buy 3,724 1,58 UBS Securities Australia Ltd Buy 3,724 1,68 UBS Securities Australia Ltd Buy 3,610 3,78 UBS Securities Australia Ltd Buy 3,624 5,78 UBS Securities Australia Ltd Buy 3,624 5,78 UBS Securities Australia Ltd Buy 3,625 1,69 UBS Securities Australia Ltd Buy 3,69 1,695 1,69 UBS Securities Australia Ltd Buy 3,69 3,67 1,69 UBS Securities Austra	3-May-14	UBS Securities Australia Ltd	Buy	0	-	Ordinan
UBS Securities Australia Ltd Buy 3,153 3.48 UBS Securities Australia Ltd Buy 3,153 3.48 UBS Securities Australia Ltd Buy 7,638 8.43 UBS Securities Australia Ltd Buy 3,152 3.48 UBS Securities Australia Ltd Buy 3,152 1,514 UBS Securities Australia Ltd Buy 1,524 1,68 UBS Securities Australia Ltd Buy 3,728 1,108 UBS Securities Australia Ltd Buy 1,524 1,68 UBS Securities Australia Ltd Buy 1,074 1,118 UBS Securities Australia Ltd Buy 1,074 1,105 UBS Securities Australia Ltd Buy 1,022 1,105 UBS Securities Australia Ltd Buy 1,295 1,54 UBS Securities Australia Ltd Buy 1,295 1,54 UBS Securities Australia Ltd Buy 3,035 3,53 UBS Securities Australia Ltd Buy 3,035 3,53 UBS Securities Australia Ltd Buy <td>3-May-14</td> <td>LIBS Securities Australia I to</td> <td>Brit</td> <td>1 540</td> <td>170</td> <td>Croling</td>	3-May-14	LIBS Securities Australia I to	Brit	1 540	170	Croling
UBS Securities Australia Ltd Buy 13,172 1,514 UBS Securities Australia Ltd Buy 7,538 843 UBS Securities Australia Ltd Buy 3,153 348 UBS Securities Australia Ltd Buy 1,524 1,667 UBS Securities Australia Ltd Buy 1,022 1,105 UBS Securities Australia Ltd Buy 1,022 1,105 UBS Securities Australia Ltd Buy 1,395 1,48 UBS Securities Australia Ltd Buy 1,002 1,105 UBS Securities Australia Ltd Buy 1,002 1,009 UBS Securities Australia Ltd Buy 3,035 3,035 UBS Securities Australia Ltd Buy 3,035 1,019 UBS Securities Australia Ltd B	2-May-14	II IRS Securities Australia I tol	(Sale	2 152	240	Section 5
UBS Securities Australia Ltd Buy 7,711 UBS Securities Australia Ltd Buy 7,638 843 UBS Securities Australia Ltd Buy 1,57 LD 1,667 UBS Securities Australia Ltd Buy 1,57 LD 1,668 UBS Securities Australia Ltd Buy 3,724 1,68 UBS Securities Australia Ltd Buy 3,724 1,68 UBS Securities Australia Ltd Buy 3,724 5,76 UBS Securities Australia Ltd Buy 3,610 3,68 UBS Securities Australia Ltd Buy 1,169 1,118 UBS Securities Australia Ltd Buy 1,295 1,65 UBS Securities Australia Ltd Buy 1,295 1,65 UBS Securities Australia Ltd Buy 3,278 3,28 UBS Securities Australia Ltd Buy 3,678 3,28 UBS Securities Australia Ltd Buy 3,678 4,28 UBS Securities Australia Ltd Buy 3,744 3,37 UBS Securities Australia Ltd Buy 3,744 </td <td>2-May-14</td> <td>I IRS Securities Australia 14d</td> <td>, and</td> <td>712 21</td> <td>1 1 1 1</td> <td>Gulliary</td>	2-May-14	I IRS Securities Australia 14d	, and	712 21	1 1 1 1	Gulliary
UBS Securities Australia Ltd Buy 3,153 348 UBS Securities Australia Ltd Buy 3,153 348 UBS Securities Australia Ltd Buy 15,120 1,667 UBS Securities Australia Ltd Buy 9,868 1,088 UBS Securities Australia Ltd Buy 1,524 1,11 UBS Securities Australia Ltd Buy 3,728 411 UBS Securities Australia Ltd Buy 1,315 1,45 UBS Securities Australia Ltd Buy 1,032 1,105 UBS Securities Australia Ltd Buy 1,032 1,105 UBS Securities Australia Ltd Buy 1,295 1,68 UBS Securities Australia Ltd Buy 1,295 1,68 UBS Securities Australia Ltd Buy 5,224 5,8 UBS Securities Australia Ltd Buy 1,395 1,69 UBS Securities Australia Ltd Buy 3,734 3,73 UBS Securities Australia Ltd Buy 3,744 3,77 UBS Securities Australia Ltd Buy	Mary 14	LIBS Cognition Acception to	bouy B	11/151	410,1	Ordinary
UBS Securities Australia Ltd Buy 15.120 1.667 UBS Securities Australia Ltd Buy 1.5,120 1.667 UBS Securities Australia Ltd Buy 1.5,120 1.667 UBS Securities Australia Ltd Buy 1.5,224 1.618 UBS Securities Australia Ltd Buy 3.728 411 UBS Securities Australia Ltd Buy 3.724 411 UBS Securities Australia Ltd Buy 3.724 411 UBS Securities Australia Ltd Buy 1.0140 1.118 UBS Securities Australia Ltd Buy 1.022 1.105 UBS Securities Australia Ltd Buy 1.022 1.00 UBS Securities Australia Ltd Buy 1.285 1.65 UBS Securities Australia Ltd Buy 3.736 3.73 UBS Securities Australia Ltd B	-Iviay-I+	ODS Securities Australia Liu	Áng	7,036	843	Ordinary
UBS Securities Australia Ltd Buy 15 120 1687 UBS Securities Australia Ltd Buy 15 120 1 687 UBS Securities Australia Ltd Buy 3,724 168 UBS Securities Australia Ltd Buy 3,724 171 UBS Securities Australia Ltd Buy 3,724 5,76 UBS Securities Australia Ltd Buy 3,610 398 UBS Securities Australia Ltd Buy 10,140 1,118 UBS Securities Australia Ltd Buy 10,022 1,105 UBS Securities Australia Ltd Buy 1,395 165 UBS Securities Australia Ltd Buy 3,278 4,28 UBS Securities Australia Ltd Buy 3,278 3,28 UBS Securities Australia Ltd Buy 3,235 1,019 UBS Securities Australia Ltd Buy 3,035 3,37 UBS Securities Australia Ltd Buy 3,035 3,37 UBS Securities Australia Ltd Buy 3,035 3,39 UBS Securities Australia Ltd Buy	-May-14	UBS Securities Australia Ltd	Buy	3,153	348	Ordinary
UBS Securities Australia Ltd Buy 15,120 1,667 UBS Securities Australia Ltd Buy 9,868 1,088 UBS Securities Australia Ltd Buy 1,224 1,667 UBS Securities Australia Ltd Buy 1,324 1,61 UBS Securities Australia Ltd Buy 1,324 1,51 UBS Securities Australia Ltd Buy 1,31 1,45 UBS Securities Australia Ltd Buy 1,0140 1,118 UBS Securities Australia Ltd Buy 1,395 1,54 UBS Securities Australia Ltd Buy 1,395 1,65 UBS Securities Australia Ltd Buy 3,62 1,09 UBS Securities Australia Ltd Buy 3,63 3,63 UBS Securities Australia Ltd Buy 3,73 1,01 UBS Securities Australia Ltd Buy 3,73 1,01 UBS Securities Australia Ltd Buy 3,73 1,01 UBS Securities Australia Ltd Buy 3,74 4,38 UBS Securities Australia Ltd Buy	-May-14	UBS Securties Australia Ltd	Buy	190	21	Ordinary
UBS Securities Australia Ltd Buy 9.868 1,088 UBS Securities Australia Ltd Buy 1,524 116 UBS Securities Australia Ltd Buy 3,728 411 UBS Securities Australia Ltd Buy 3,610 398 UBS Securities Australia Ltd Buy 10,140 1,118 UBS Securities Australia Ltd Buy 1,022 1,105 UBS Securities Australia Ltd Buy 1,495 1,65 UBS Securities Australia Ltd Buy 1,495 1,65 UBS Securities Australia Ltd Buy 5,25 58 UBS Securities Australia Ltd Buy 3,878 4,28 UBS Securities Australia Ltd Buy 3,878 4,28 UBS Securities Australia Ltd Buy 3,878 4,28 UBS Securities Australia Ltd Buy 3,74 4,387 UBS Securities Australia Ltd Buy 3,74 4,387 UBS Securities Australia Ltd Buy 3,74 4,387 UBS Securities Australia Ltd Buy	9-May-14	UBS Securities Australia Ltd	Buy	15,120	1,667	Ordinary
UBS Securities Australia Ltd Buy 1,524 168 UBS Securities Australia Ltd Buy 3,728 411 UBS Securities Australia Ltd Buy 5,224 5/6 UBS Securities Australia Ltd Buy 1,0140 1,118 UBS Securities Australia Ltd Buy 1,0740 1,118 UBS Securities Australia Ltd Buy 1,022 1,105 UBS Securities Australia Ltd Buy 1,295 1,64 UBS Securities Australia Ltd Buy 1,395 1,64 UBS Securities Australia Ltd Buy 3,87B 42B UBS Securities Australia Ltd Buy 3,67B 42B UBS Securities Australia Ltd Buy 3,73B 42B UBS Securities Australia Ltd Buy	3-May-14	UBS Securities Australia Ltd	Buy	898'6	1,088	Ordinary
UBS Securities Australia Ltd Buy 3.728 411 UBS Securities Australia Ltd Buy 1,315 145 UBS Securities Australia Ltd Buy 5,224 5,76 UBS Securities Australia Ltd Buy 10,140 1,118 UBS Securities Australia Ltd Buy 10,022 1,105 UBS Securities Australia Ltd Buy 1,395 1,62 UBS Securities Australia Ltd Buy 5,25 58 UBS Securities Australia Ltd Buy 5,25 58 UBS Securities Australia Ltd Buy 5,25 1,89 UBS Securities Australia Ltd Buy 3,73 1,019 UBS Securities Australia Ltd Buy 3,73 1,019 UBS Securities Australia Ltd Buy 3,73 1,019 UBS Securities Australia Ltd Buy 3,74 4,337 UBS Securities Australia Ltd Buy 3,74 4,337 UBS Securities Australia Ltd Buy 3,74 4,339 UBS Securities Australia Ltd Buy	9-May-14	UBS Securities Australia Ltd	Buy	1,524	168	Ordinary
UBS Securities Australia Ltd Buy 1,315 145 UBS Securities Australia Ltd Buy 5,224 576 UBS Securities Australia Ltd Buy 10,140 1,118 UBS Securities Australia Ltd Buy 1,0022 1,105 UBS Securities Australia Ltd Buy 1,295 1,65 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 3,878 4,28 UBS Securities Australia Ltd Buy 3,678 4,28 UBS Securities Australia Ltd Buy 3,74 4,28 UBS Securities Australia Ltd Buy 3,74 4,38 UBS Securities Australia Ltd Buy 3,7	3-May-14	UBS Securities Australia Ltd	Buy	3,728	411	Ordinary
UBS Securities Australia Ltd Buy 5,224 576 UBS Securities Australia Ltd Buy 10,140 1,105 UBS Securities Australia Ltd Buy 10,022 1,105 UBS Securities Australia Ltd Buy 1,395 1,54 UBS Securities Australia Ltd Buy 5,25 58 UBS Securities Australia Ltd Buy 5,25 58 UBS Securities Australia Ltd Buy 3,878 4,78 UBS Securities Australia Ltd Buy 3,735 3,93 UBS Securities Australia Ltd Buy 3,735 3,93 UBS Securities Australia Ltd Buy 9,736 1,019 UBS Securities Australia Ltd Buy 9,736 1,019 UBS Securities Australia Ltd Buy 9,736 1,019 UBS Securities Australia Ltd Buy 9,232 1,019 UBS Securities Australia Ltd Buy 3,744 4,387 UBS Securities Australia Ltd Buy 3,025 5,88 UBS Securities Australia Ltd Buy	3-May-14	UBS Securities Australia Ltd	Buv	1.315	145	Ordinary
UBS Securities Australia Ltd Buy 3,610 3,98 UBS Securities Australia Ltd Buy 10,140 1,118 UBS Securities Australia Ltd Buy 1,022 1,105 UBS Securities Australia Ltd Buy 1,495 165 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 3,878 4,28 UBS Securities Australia Ltd Buy 3,679 1,079 UBS Securities Australia Ltd Buy 3,776 1,079 UBS Securities Australia Ltd Buy 3,776 1,079 UBS Securities Australia Ltd Buy 3,746 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy <td>9-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>5.224</td> <td>576</td> <td>Ordinary</td>	9-May-14	UBS Securities Australia Ltd	Buv	5.224	576	Ordinary
UBS Securities Australia Ltd Buy 10,140 1,118 UBS Securities Australia Ltd Buy 1,022 1,105 UBS Securities Australia Ltd Buy 1,395 165 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 3,878 428 UBS Securities Australia Ltd Buy 3,73 39 UBS Securities Australia Ltd Buy 9,735 1,019 UBS Securities Australia Ltd Buy 9,735 1,019 UBS Securities Australia Ltd Buy 9,736 1,019 UBS Securities Australia Ltd Buy 9,736 1,019 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 5,659 5,68 UBS Securities Australia Ltd Buy 5,669 40a UBS Securities Australia Ltd Buy	3-May-14	LIBS Securities Australia 11d	8	3,610	a a a	Ordinary
USS Securities Australia Ltd Buy 1,105 UBS Securities Australia Ltd Buy 1,395 1,105 UBS Securities Australia Ltd Buy 1,395 165 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 353 39 UBS Securities Australia Ltd Buy 373 1,019 UBS Securities Australia Ltd Buy 373 1,019 UBS Securities Australia Ltd Buy 374 4,337 UBS Securities Australia Ltd Buy 374 4,337 UBS Securities Australia Ltd Buy 30,35 1,019 UBS Securities Australia Ltd Buy 30,25 1,019 UBS Securities Australia Ltd Buy 80,7 99	2-May-14	PS Congrities Australia 144	600	10.140	1110	ording.
UBS Securities Australia Ltd Buy 1,022 1,103 UBS Securities Australia Ltd Buy 1,495 165 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 3,878 428 UBS Securities Australia Ltd Buy 3,678 35 UBS Securities Australia Ltd Buy 3,776 1,079 UBS Securities Australia Ltd Buy 9,232 1,019 UBS Securities Australia Ltd Buy 3,744 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 5,055 5,659 UBS Securities Australia Ltd Buy 5,055 5,659 UBS Securities Australia Ltd Buy 5,650 404 UBS Securities Australia Ltd Buy	May 14	I DC Countries Australia Ltd	bug b	10,140	1,10	Odinary
UBS Securities Australia Ltd Buy 1,399 158 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 3,878 428 UBS Securities Australia Ltd Buy 3,635 39 UBS Securities Australia Ltd Buy 3,735 39 UBS Securities Australia Ltd Buy 9,736 1,019 UBS Securities Australia Ltd Buy 9,736 4,387 UBS Securities Australia Ltd Buy 9,237 1,019 UBS Securities Australia Ltd Buy 5,659 774 UBS Securities Australia Ltd Buy 5,659 56 UBS Securities Australia Ltd Buy 5,650 40a UBS Securities Australia Ltd Buy 5,650 40a UBS Securities Australia Ltd Buy	1	The Committee Australia Line	500	10,022	1,103	Clallay
UBS Securities Australia Ltd Buy 1,495 TbS UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 525 58 UBS Securities Australia Ltd Buy 3878 428 UBS Securities Australia Ltd Buy 3,035 335 UBS Securities Australia Ltd Buy 9,776 1,079 UBS Securities Australia Ltd Buy 3,746 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 30,251 3,387 UBS Securities Australia Ltd Buy 30,251 3,387 UBS Securities Australia Ltd Buy 5,055 5,88 UBS Securities Australia Ltd Buy 5,660 4,04 UBS Securities Australia Ltd Buy	-Iviay-14	Opp Securities Australia Ltd	anà	CRE'I	154	Ordinary
UBS Securities Australia Ltd Buy 5.25 SB UBS Securities Australia Ltd Buy 5.25 SB UBS Securities Australia Ltd Buy 3,878 4.28 UBS Securities Australia Ltd Buy 3,035 335 UBS Securities Australia Ltd Buy 9,726 1,079 UBS Securities Australia Ltd Buy 9,232 1,019 UBS Securities Australia Ltd Buy 3,746 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 5,055 5,055 UBS Securities Australia Ltd Buy 5,055 5,055 UBS Securities Australia Ltd Buy 8,07 99 UBS Securities Australia Ltd Buy 5,055 5,055 UBS Securities Australia Ltd Buy 5,055 5,055 UBS Securities Australia Ltd Buy 5,055 5,055 UBS Securities Australia Ltd Buy	P-Ividy-14	ODS SECULINES AUSTRAINA LIG	any	1,490	cal c	Ordinary
UBS Securities Australia Ltd Buy 3.52 3.8 UBS Securities Australia Ltd Buy 3.63 4.38 UBS Securities Australia Ltd Buy 3.635 3.95 UBS Securities Australia Ltd Buy 9.776 1.019 UBS Securities Australia Ltd Buy 9.732 1.019 UBS Securities Australia Ltd Buy 9.734 4.387 UBS Securities Australia Ltd Buy 3.744 4.387 UBS Securities Australia Ltd Buy 3.0,251 3.399 UBS Securities Australia Ltd Buy 5.055 5.8 UBS Securities Australia Ltd Buy 5.055 5.8 UBS Securities Australia Ltd Buy 5.055 5.9 UBS Securities Australia Ltd Buy 5.055 5.05 UBS Securities Australia Ltd Buy	-May-14	UBS Securities Australia Ltd	Buy	575	80	Ordinary
UBS Securities Australia Ltd Buy 35.78 428 UBS Securities Australia Ltd Buy 36.35 33.9 UBS Securities Australia Ltd Buy 3,035 33.5 UBS Securities Australia Ltd Buy 9,776 1,079 UBS Securities Australia Ltd Buy 9,726 1,079 UBS Securities Australia Ltd Buy 39,746 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 6,469 714 UBS Securities Australia Ltd Buy 6,469 714 UBS Securities Australia Ltd Buy 5,055 5,86 UBS Securities Australia Ltd Buy 5,665 5,98 UBS Securities Australia Ltd Buy 5,663 404 UBS Securities Australia Ltd Buy 5,663 6,55 UBS Securities Australia Ltd Buy 5,663 6,55	3-May-14	UBS Securities Australia Ltd	Buy	525	28	Ordinary
UBS Securities Australia Ltd Buy 353 39 UBS Securities Australia Ltd Buy 3/035 335 UBS Securities Australia Ltd Buy 9,776 1,079 UBS Securities Australia Ltd Buy 9,232 1,019 UBS Securities Australia Ltd Buy 3,144 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 5,659 774 UBS Securities Australia Ltd Buy 5,659 774 UBS Securities Australia Ltd Buy 5,659 40a UBS Securities Australia Ltd Buy 5,650 40a UBS Securities Australia Ltd Buy 5,653 6,55 UBS Securities Australia Ltd Buy 5,653 6,55 UBS Securities Australia Ltd Buy 5,653 6,55	9-May-14	UBS Securities Australia Ltd	Buy	3,878	428	Ordinary
UBS Securities Australia Ltd Buy 3,035 335 UBS Securities Australia Ltd Buy 9,776 1,079 UBS Securities Australia Ltd Buy 9,232 1,079 UBS Securities Australia Ltd Buy 33,746 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 6,055 5,88 UBS Securities Australia Ltd Buy 6,055 5,88 UBS Securities Australia Ltd Buy 897 99 UBS Securities Australia Ltd Buy 5,055 50	3-May-14	UBS Securities Australia Ltd	Buy	353	39	Ordinary
UBS Securities Australia Ltd Buy 9,776 1,079 UBS Securities Australia Ltd Buy 9,232 1,019 UBS Securities Australia Ltd Buy 3,746 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 6,469 714 UBS Securities Australia Ltd Buy 6,469 714 UBS Securities Australia Ltd Buy 5,055 5,86 UBS Securities Australia Ltd Buy 897 99 UBS Securities Australia Ltd Buy 5,665 404 UBS Securities Australia Ltd Buy 5,663 404 UBS Securities Australia Ltd Buy 5,663 6,25	3-May-14	UBS Securities Australia Ltd	Buy	3,035	335	Ordinary
UBS Securities Australia Ltd Buy 9,232 1,019 UBS Securities Australia Ltd Buy 31,44 347 UBS Securities Australia Ltd Buy 30,251 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 5,659 774 UBS Securities Australia Ltd Buy 5,659 598 UBS Securities Australia Ltd Buy 5,650 40a UBS Securities Australia Ltd Buy 5,670 40a UBS Securities Australia Ltd Buy 5,670 40a UBS Securities Australia Ltd Buy 5,673 5,65 UBS Securities Australia Ltd Buy 5,683 6,55	3-May-14	UBS Securities Australia Ltd	Buy	922'6	1,079	Ordinary
UBS Securities Australia Ltd Buy 3,144 347 UBS Securities Australia Ltd Buy 39,746 4,387 UBS Securities Australia Ltd Buy 30,251 3,399 UBS Securities Australia Ltd Buy 6,469 714 UBS Securities Australia Ltd Buy 897 99 UBS Securities Australia Ltd Buy 897 99 UBS Securities Australia Ltd Buy 5,055 404 UBS Securities Australia Ltd Buy 5,075 5,075 UBS Securities Australia Ltd Buy 5,075 5,075 UBS Securities Australia Ltd Buy 5,075 5,075 UBS Securities Australia Ltd Buy 5,063 6,25	9-May-14	UBS Securities Australia Ltd	Buy	9,232	1,019	Ordinary
UBS Securities Australia Ltd Buy 39,746 4,387 UBS Securities Australia Ltd Buy 30,251 3,339 UBS Securities Australia Ltd Buy 6,469 714 UBS Securities Australia Ltd Buy 897 5,655 558 UBS Securities Australia Ltd Buy 897 99 UBS Securities Australia Ltd Buy 5,605 40a UBS Securities Australia Ltd Buy 5,605 40a UBS Securities Australia Ltd Buy 5,675 5,655 UBS Securities Australia Ltd Buy 5,663 6,25	9-Mav-14	UBS Securities Australia Ltd	Buy	3.144	347	Ordinary
USE Securities Australia Ltd Buy 30,511 3,339 USE Securities Australia Ltd Buy 6,469 714 USE Securities Australia Ltd Buy 5,055 58 USE Securities Australia Ltd Buy 897 99 USE Securities Australia Ltd Buy 3,660 40a USE Securities Australia Ltd Buy 5,65 40a USE Securities Australia Ltd Buy 5,65 6,55 USE Securities Australia Ltd Buy 5,663 6,55	-May-14	I IBS Securities Australia Ltd	A 6	30.746	1 307	Coding
USD Securities Australia Ltd Buy 5,055 73.9 UBS Securities Australia Ltd Buy 6,469 714 UBS Securities Australia Ltd Buy 897 99 UBS Securities Australia Ltd Buy 887 99 UBS Securities Australia Ltd Buy 5,055 404 UBS Securities Australia Ltd Buy 5,075 5,075	MAN, 1A	I IDC Consister Assessment Ltd	600	200251	000,0	Cicilian y
Usb Securities Australia Ltd Buy 0,403 714 UBS Securities Australia Ltd Buy 5,055 558 558 UBS Securities Australia Ltd Buy 897 99 UBS Securities Australia Ltd Buy 5,050 404 UBS Securities Australia Ltd Buy 5,053 5,663 UBS Securities Australia Ltd Buy 5,663 6,25		tibe comittee Australia Ltd	ond o	162,06	2,239	Ordinary
UBS Securities Australia Ltd Buy 5,055 5,88 UBS Securities Australia Ltd Buy 897 99 UBS Securities Australia Ltd Buy 3,660 404 UBS Securities Australia Ltd Buy 507 56 UBS Securities Australia Ltd Buy 5,663 625	2-Ividy-14	Ubb Securities Australia Ltd	Buy	6,409	/14	Ordinary
UBS Securities Australia Ltd Buy 897 99 UVBS Securities Australia Ltd Buy 3,660 404 UBS Securities Australia Ltd Buy 507 56 UBS Securities Australia Ltd Buy 5,663 625	-May-14	UBS Securities Australia Ltd	Buy	9,055	258	Ordinary
UBS Securities Australia Ltd Buy 3.660 40a UBS Securities Australia Ltd Buy 5.07 5.6 UBS Securities Australia Ltd Buy 5,663 6,25	9-May-14	UBS Securities Australia Ltd	Виу	897	66	Ordinary
UBS Securities Australia Ltd Buy 507 56 UBS Securities Australia Ltd Buy 5,663 625	9-May-14	UBS Securities Australia Ltd	Buy	3,660	404	Ordinary
UBS Securities Australia Ltd Buy 5,663 6.25	9-May-14	UBS Securities Australia Ltd	Виу	507	95	Ordinary
Social Control of the	2-May-14					,
11.6 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7		II IRC Somerition Ametralia I to	Ren	5 663	675	Ordinar,

9-May-14				477	Celler
The state of	I IRS Securities Australia 11d	Ris	3 448	381	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	38.082	4 208	Ordinary
9-May-14	I BS Securities Australia I to	Buy	4 851	753	Ordinary
t (a)	COS Securities Australia Liu	ing.	1,007	occ.	Cidinary
9-May-14	Obs Securities Australia Ltd	ana -	C.S.	SO2	Orginary
y-May-14	UBS Securities Australia Ltd	Buy	7,082	730	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	25,340	2,800	Ordinary
May-14	UBS Securities Australia Ltd	Buy	1,077	119	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	4,597	208	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,059	117	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	9.055	1.000	Ordinary
19-Mav-14	UBS Securities Australia Ltd	Buv	8.150	CUG	Ordinary
May 14	+	, Big	31 3/8	2 462	Ordinan
	The Commission Assessed the Commission of the Co		170,10	204,0	Oldinary
Mdy-14	╅	Buy	5,304	760	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	4,766	276	Ordinary
9-May-14		Buy	906	100	Ordinary
May-14		Buy	298	99	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	3,434	379	Ordinary
May-14		Buy	3.180	351	Ordinary
May-14		Buy	1,848	202	Ordinary
9-May-14		BIN	16,226	1 701	Ordinary
O Mar 14	I BC Coourities Australia 140	Bin	233 (101	Ordinary
IVIGY 14	ODS SECULIES AUSTRAIN LIG	ind a	(00's	5	Ciclialy
3-IMIGN-14	Coo Securities Australia Liu	ea,	050'1	171	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,250	138	Ordinary
May-14	UBS Securities Australia Ltd	Buy	23,792	2,626	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	190	21	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,060	117	Ordinary
9-May-14	I IBS Securities Australia I td	Biv	805.5	375	Ordinary
N VIEW OI	IDO Con printers Americal 14	200	000 /	S	Ordinan
viay i	COO Securities Australia Lid	(ng	055,4	200	Cidinary
9-May- ₹4	UBS Securities Australia Ltd	any	3,153	348	Ordinary
May-14	UBS Securities Australia Ltd	Buy	31,710	3,500	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	19,391	265	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,033	114	Ordinary
Q.M.34.14	I IRC Securities Australia 14d	B	1 1 1 2 3	121	Ordinan
O May 14	Lat Constant American	(5)	263 C	1900	Ordinal y
	מסח הבימו ווופיז שתזוו מוום דיות	62	000,2	167	0.00
9-May-14	Obs securities Australia Ltd	buy	069'6	430	Ordinary
May-14	UBS Securities Australia Ltd	Buy	876	108	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	16,498	1,819	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buv	14.757	1.627	Ordinary
0. May 14	1 BS Counities Australia 144	Biac	10.038	1 208	Ordinan
lviay- i-	Ť	, and	956'0'	007,1	Cidinary
May-14	4	Buy	906	201	Ordinary
May-14	UBS Securities Australia Ltd	Buy	6,371	704	Ordinary
19-May-14	f	Buy	1,167	129	Ordinary
9-Mav-14	┢	Buv	1.041	115	Ordinary
1	I IDO Commission Australia Lea	(2)	3701	700	Carrier C
19-May-14	UBS Securities Australia Ltd	buy	1,840	70d	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,326	257	Ordinary
May-14	UBS Securities Australia Ltd	Buy	13,586	1,502	Ordinary
19-Mav-14	UBS Securities Australia Ltd	Buv	12,227	1.351	Ordinary
9-May-14	I IBS Securities Australia I to	Bin	29 765	3 280	Ordina
,	Call Control A control of the		001/03	2,400	Con and
9-May-14	UBS Securities Australia Ltd	pay	916	'n	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	11,032	1,219	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	4,950	547	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	980'E	341	Ordinary
9-May-14	LIBS Securities Australia Ltd	Buy	1 466	162	Ordinary
24.0	DO CO Contraction Assessment and	:::a	45 040	20.	Constraint of
Mdy-14	Coo Securities Australia Liu	ons	010,61	1,747	Crainary
19-May-14	UBS Securities Australia Ltd	Buy	3,349	370	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,928	213	Ordinary
May-14	UBS Securities Australia Ltd	Buy	10,507	1,161	Ordinary
19-Mav-14	UBS Securities Australia Ltd	Buv	2.037	225	Ordinary
1,	Control of the state of the sta	600	500	00,	The second
9-May-14	UBS Securities Australia Ltd	hoy	ans	3	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,268	140	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,613	178	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	3,135	346	Ordinary
May-14	I IBC Cognitities Australia I to	Riv	1 98/1	210	Ordinary
Niay	ייייי ייייי אייייי איייייי אייייי	, con		1117	555
19-May-14	I IDS Convition Australia Ltd		0 0 0		, , , , , , , , , , , , , , , , , , , ,
-	מבים ונובי שחפון מוומ בנת	BUÿ	525	85	Ordinary

Big Securities Australia Ltd Bly 17,220 17,220 18, 18, 18, 18, 18, 18, 18, 18, 18, 18,	20-May-14	I IBC Con critical Assessed 1404				26050
List Securities Australia Ltd Bay 6.75 6.75 List Securities Australia Ltd Sell 1.97 6.75 6.75 List Securities Australia Ltd Sell 1.37 7.74 7.74 7.74 List Securities Australia Ltd Entre Into Derivative (1.00 1.73 7.74	-Mav-14	I IRS Securities Australia 11d	2 3	051,651	(910,55)	City of
Bay 1,295 1,294 1,295	1-May-14	UBS Securities Australia Ltd	Buv	421, 125	46.751	Ordinary
Securities Australia Ltd Buy 1,23,592 1,3,7592 1,3,7593	-Mav-14	LIBS Securities Australia I td	Riny	91 579	2000	Ordinary Ordinary
Bis Securities Australial Ltd First Into Derivative 67,172 (1,452) Bis Securities Australial Ltd Firster Into Derivative 43,386 50,018 Bis Securities Australial Ltd Firster Into Derivative 1,471 2,385 Bis Securities Australial Ltd Firster Into Derivative 1,471 2,385 Bis Securities Australial Ltd Buy 4,984 5,380 Bis Securities Australial Ltd Buy 2,772 300 Bis Securities Australial Ltd Buy 2,772 300 Bis Securities Australial Ltd Buy 2,576 5,00 Bis Securities Australial Ltd Buy 2,576 5,00 Bis Securities Australial Ltd Buy 2,576 5,00 Bis Securities Australial Ltd Buy 2,526 2,60 Bis Securities Australial Ltd Buy 2,52 2,60 Bis Securities Australial Ltd Buy 2,52 2,50 Bis Securities Australial Ltd Buy 2,52 2,50 Bis Securities Australial Ltd Buy 2,52 2,	-May-14	I IRS Securities Australia Ltd	100	100 501	(40,500)	Ordinary
DES Securities Autatrials Ltd BMY 45/33 DES Securities Autatrials Ltd Enter into Derivative NA (500) DES Securities Autatrials Ltd Enter into Derivative NA (500) DES Securities Autatrials Ltd Bluy 48,886 2,580 DES Securities Autatrials Ltd Bluy 4,888 2,580 DES Securities Autatrials Ltd Bluy 4,888 2,580 DES Securities Autatrials Ltd Bluy 4,522 80 DES Securities Autatrials Ltd Bluy 4,522 80 DES Securities Autatrials Ltd Bluy 1,137 1,137 DES Securities Autatrials Ltd Bluy 1,522 1,522 DES Securities Autatrials Ltd Bluy 1,527 2,526 DES Securities Autatrials Ltd Bluy 1,527 3,500 DES Securities		CIDS Comments A contract of the	i c	286,621	(13,734)	Ordinary
DES Securities Automitation Bluy previotive 49.5 Mobile 50.00 BOS Securities Automitation Einter Into Derivative NAM (50.00) BOS Securities Automagnement Bluy 2.175.14 2.385 BOS Socialities Automagnement Bluy 2.372 3.80 BOS Socialities Automagnement Automagnement Bluy 2.372 3.80 BOS Socialities Automagnement Bluy 8.04 4.30 BOS Socialities Automagnement Automagnement Automagnement Automagnement Automagnement Bluy Bluy 5.569 3.00 BOS Socialities Automagnement Bluy Bluy 1.543 1.541 1.542 BOS Socialities Automagnement Bluy Bluy 2.328 1.529 1.529 BOS Socialities Automagnement Bluy	100	CO Securities Australia Liu	nen V	2/1//0	(7,455)	Ordinary
Box Securities Autorials Ltd Einter Into Derivative NAA (550)	-Ivay-14	COS Securities Australia LIO	Buy	453,80b	50,648	Ordinary
UBS Scornine Australia Int of Brite into Derivative (2) INA (2006) (1	-May-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(321)	Options
UBS Colouries Australia Ltd Buy 21,514 2,586 UBS Colouries Australia Ltd Buy 21,512 5.30 UBS Colouries Australia Ltd Buy 4,199 4,65 UBS Securities Australia Ltd Buy 7,322 810 UBS Securities Australia Ltd Buy 7,522 810 UBS Securities Australia Ltd Buy 7,522 81 UBS Securities Australia Ltd Buy 7,526 1,570 UBS Securities Australia Ltd Buy 7,541 1,520 UBS Securities Australia Ltd Buy 7,542 1,528 UBS Securities Australia Ltd Buy 7,542 1,542 UBS Securities Australia Ltd Buy	-May-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	(200)	Options
UBS Countries Australia Ltd Blyw 48,48B \$380 UBS Chebla Acses Mériagement Self 5,380 (0.00) UBS Chebla Acses Mériagement Buy 7,372 580 UBS Wealth Management Australia Ltd Buy 7,322 610 UBS Wealth Management Australia Ltd Buy 4,199 465 UBS Securities Australia Ltd Buy 4,199 47,90 UBS Securities Australia Ltd Buy 45,00 40,90 UBS Securities Australia Ltd Buy 45,00 40,10 UBS Securities Australia Ltd Buy 45,00 40,40 UBS Securities Australia Ltd Buy 5,60 5,20 UBS Securities Australia Ltd Buy 5,60 5,20 UBS Securities Australia Ltd Buy 5,00 5,20 UBS Securities Australia Ltd Buy 5,00 5,20 UBS Securities Australia Ltd Buy 5,00 5,20 UBS Securities Australia Ltd Buy 1,60 5,24 UBS Securities Australia Ltd <td< td=""><td>-May-14</td><td>UBS Global Asset Management</td><td>Buy</td><td>21,514</td><td>2,385</td><td>Ordinary</td></td<>	-May-14	UBS Global Asset Management	Buy	21,514	2,385	Ordinary
UBS Vehigh Management Activation Self 3.699 (40,0) UBS Weight Management Activation Bay 2,732 80 UBS Weight Management Activation and Management Activation Bay 7,322 81 UBS Securities Australia Ltd Buy 7,324 1,47 UBS Securities Australia Ltd Buy 1,024 1,47 UBS Securities Australia Ltd Buy 2,736 5,000 UBS Securities Australia Ltd Buy 2,736 1,737 UBS Securities Australia Ltd Buy 2,736 1,737 UBS Securities Australia Ltd Buy 2,736 1,737 UBS Securities Australia Ltd Buy 2,738 1,736 UBS Securities Australia Ltd Buy 2,738 1,737 UBS Securities Australia Ltd Buy 2,738 1,737 UBS Securities Australia Ltd Buy 2,738 1,244 UBS Securities Australia Ltd Buy 2,738 2,244 UBS Securities Australia Ltd Buy 2,738 3,244 UBS Securit	-May-14	UBS Global Asset Management	Buy	48,488	5.380	Ordinary
UBS Vectorities Australia Ltd Bly 2,772 900 UBS Securities Australia Ltd Buy 7,322 810 UBS Securities Australia Ltd Buy 7,322 810 UBS Securities Australia Ltd Buy 45,000 5,000 UBS Securities Australia Ltd Buy 45,000 5,000 UBS Securities Australia Ltd Buy 45,000 5,000 UBS Securities Australia Ltd Buy 5,600 5,000 UBS Securities Australia Ltd Buy 5,600 5,000 UBS Securities Australia Ltd Buy 5,600 5,000 UBS Securities Australia Ltd Buy 5,000 5,000 UBS Securities Australia Ltd <th< td=""><td>J-May-14</td><td>UBS Global Asset Management</td><td>Sell</td><td>3 639</td><td>(404)</td><td>Ordinary</td></th<>	J-May-14	UBS Global Asset Management	Sell	3 639	(404)	Ordinary
UBS Securities Australia Ltd BMy 1,522 610 LDBS Securities Australia Ltd BMy 1,0244 1,670 LDBS Securities Australia Ltd BMy 1,0244 1,570 LDBS Securities Australia Ltd BMy 223,367 2,900 LDBS Securities Australia Ltd BMy 1,643,34 1,670 LDBS Securities Australia Ltd BMy 1,643,34 1,671 LDBS Securities Australia Ltd BMy 1,673 3,000 LDBS Securities Australia Ltd BMy 2,005,553 7,265 LDBS Securities Australia Ltd BMy 1,673 1,274 LDBS Securities Australia Ltd BMy 1,673 1,244 LDBS Securities Australia Ltd BMy 1,674 1,282 LDBS Securities Australia Ltd BMy 1,674 1,282 LDBS Sec	-Mav-14	UBS Wealth Management Australia Ltd	Bliv	2712	005	Ording
UBS Securities Australia Ltd BMy 4,122.8 466 UBS Securities Australia Ltd BMy 4,102.44 1,137 UBS Securities Australia Ltd BMy 45,000 24,901 UBS Securities Australia Ltd BMy 1,234 1,137 UBS Securities Australia Ltd BMy 1,64,324 1,137 UBS Securities Australia Ltd BMy 1,64,324 1,404 UBS Securities Australia Ltd BMy 3,640 1,71 UBS Securities Australia Ltd BMy 3,640 1,71 UBS Securities Australia Ltd BMy 2,05,633 2,000 UBS Securities Australia Ltd BMy 2,05,633 2,000 UBS Securities Australia Ltd BMy 2,05,633 3,000 UBS Securities Australia Ltd BMy 7,1,75 8,338 UBS Securities Australia Ltd BMy 7,1,75 8,338 UBS Securities Australia Ltd BMy 7,1,75 1,2,20 UBS Securities Australia Ltd BMy 7,1,75 1,2,20 UBS Securities Aust	-May-14	I IRS Wealth Management Asstratio 11d	600	2,1,12	0.00	Grainary
UBS Securities Australia Ltd BMY 10,244 1,470 UBS Securities Australia Ltd BMY 223,367 24,901 UBS Securities Australia Ltd BMY 223,367 24,901 UBS Securities Australia Ltd BMY 1,414 1,570 UBS Securities Australia Ltd BMY 16,1324 1,670 UBS Securities Australia Ltd BMY 16,1324 1,670 UBS Securities Australia Ltd BMY 3,640 1,671 UBS Securities Australia Ltd BMY 205,529 7,265 UBS Securities Australia Ltd BMY 205,529 7,265 UBS Securities Australia Ltd BMY 205,529 7,265 UBS Securities Australia Ltd BMY 1,270 1,570 UBS Securities Australia Ltd BMY 1,274 1,282 UBS Securities Australia L	May 14	LIDE Committee A metallic ted	Sec. of	226,	0 0	O'GINAL S
UBS Securities Australia Ltd Buy 41,024 1,024 UBS Securities Australia Ltd Buy 45,024 5,000 UBS Securities Australia Ltd Buy 164,224 1,270 UBS Securities Australia Ltd Buy 164,224 18,70 UBS Securities Australia Ltd Buy 3,600 404 UBS Securities Australia Ltd Buy 3,600 404 UBS Securities Australia Ltd Buy 3,600 404 UBS Securities Australia Ltd Buy 2,05,303 3,200 UBS Securities Australia Ltd Buy 2,05,303 3,200 UBS Securities Australia Ltd Buy 2,05,303 3,200 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 75,125 8,238 UBS Securities Australia Ltd<	Mary 14	CDS Securities Australia Ltd	buy	4,199	466	Ordinary
UBS Securities Australia Ltd Buy 25,050 5,000 UBS Securities Australia Ltd Buy 223,367 24,791 UBS Securities Australia Ltd Buy 16,1224 18,791 UBS Securities Australia Ltd Buy 3,640 404 UBS Securities Australia Ltd Buy 3,640 404 UBS Securities Australia Ltd Buy 3,640 3,600 UBS Securities Australia Ltd Buy 205,559 7,261 UBS Securities Australia Ltd Buy 205,503 7,265 UBS Securities Australia Ltd Buy 205,503 3,000 UBS Securities Australia Ltd Buy 20,503 3,000 UBS Securities Australia Ltd Buy 1,548 1,549 UBS Securities Australia Ltd Buy 1,548 1,573 UBS Securities Australia Ltd Buy 2,014 1,257 UBS Securities Australia Ltd Buy 2,014 1,258 UBS Securities Australia Ltd Buy 2,014 1,259 UBS Securities Australia Ltd	+I -keini	Obs securities Australia Ltd	Buy	10,244	1,13/	Ordinary
UBS Securities Australia Ltd Buy 2.23.367 24.731 UBS Securities Australia Ltd Buy 1.47.46 1.570 UBS Securities Australia Ltd Buy 1.64.324 1.67.01 UBS Securities Australia Ltd Buy 5.64.00 4.04 UBS Securities Australia Ltd Buy 5.64.00 4.04 UBS Securities Australia Ltd Buy 5.64.00 4.04 UBS Securities Australia Ltd Buy 27.03.03 3.000 UBS Securities Australia Ltd Buy 27.03.03 3.000 UBS Securities Australia Ltd Buy 2.05.63 2.54.04 UBS Securities Australia Ltd Buy 7.51.12 1.38 UBS Securities Australia Ltd Buy 7.51.12 1.8.4 UBS Securities Australia Ltd Buy 7.51.12 1.2.7 UBS Securities Australia Ltd Buy 7.51.12 1.2.2 UBS Securities Australia Ltd Buy 7.51.12 1.2.2 UBS Securities Australia Ltd Buy 5.00 5.00 UBS Securiti	- 1	UBS Securities Australia Ltd	Buy	45,050	2,000	Ordinary
UBS Securities Australia Ltd Buy 14,146 1,570 UBS Securities Australia Ltd Buy 1,4146 1,570 UBS Securities Australia Ltd Buy 1,64324 1,523 UBS Securities Australia Ltd Buy 66,359 7,361 UBS Securities Australia Ltd Buy 66,356 7,365 UBS Securities Australia Ltd Buy 20,565 2,2815 UBS Securities Australia Ltd Buy 38,793 4,306 UBS Securities Australia Ltd Buy 1,688 1,629 UBS Securities Australia Ltd Buy 1,888 1,538 UBS Securities Australia Ltd Buy 1,387 3,500 UBS Securities Australia Ltd Buy 1,387 3,540 UBS Securities Australia Ltd Buy 1,387 1,244 UBS Securities Australia Ltd Buy 1,387 1,244 UBS Securities Australia Ltd Buy 1,387 1,244 UBS Securities Australia Ltd Buy 1,244 2,244 UBS Securities Australia Ltd	•	UBS Securities Australia Ltd	Buy	223,367	24,791	Ordinary
UBS Securities Australia Ltd Buy 64,324 10.1 UBS Securities Australia Ltd Buy 164,324 16,328 UBS Securities Australia Ltd Buy 5,640 404 UBS Securities Australia Ltd Buy 66,359 7,365 UBS Securities Australia Ltd Buy 205,563 22,815 UBS Securities Australia Ltd Buy 200,563 22,815 UBS Securities Australia Ltd Buy 200,563 22,815 UBS Securities Australia Ltd Buy 200,501 5,549 UBS Securities Australia Ltd Buy 1,388 1,544 UBS Securities Australia Ltd Buy 1,43,256 1,579 UBS Securities Australia Ltd Buy 1,547 1,18 UBS Securities Australia Ltd Buy 1,547 1,18 UBS Securities Australia Ltd Buy 1,547 1,18 UBS Securities Australia Ltd Buy 2,014 2,014 UBS Securities Australia Ltd Buy 2,014 3,06 UBS Securities Australia Ltd <td></td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>14,146</td> <td>1,570</td> <td>Ordinary</td>		UBS Securities Australia Ltd	Buy	14,146	1,570	Ordinary
UBS Securities Australia Ltd Buy 164,324 18,238 UBS Securities Australia Ltd Buy 3,640 404 UBS Securities Australia Ltd Buy 6,359 7,365 UBS Securities Australia Ltd Buy 205,563 22,815 UBS Securities Australia Ltd Buy 205,563 22,815 UBS Securities Australia Ltd Buy 38,737 4,306 UBS Securities Australia Ltd Buy 11,288 16,297 UBS Securities Australia Ltd Buy 11,288 16,297 UBS Securities Australia Ltd Buy 1,388 15,44 UBS Securities Australia Ltd Buy 1,388 15,44 UBS Securities Australia Ltd Buy 1,647 128 UBS Securities Australia Ltd Buy 1,647 128 UBS Securities Australia Ltd Buy 1,647 1,28 UBS Securities Australia Ltd Buy 1,647 1,28 UBS Securities Australia Ltd Buy 1,647 1,28 UBS Securities Australia Ltd		UBS Securities Australia Ltd	Buy	6	-	Ordinary
UBS Securities Australie Ltd Buy 3,640 4,04 UBS Securities Australie Ltd Buy 66,359 7,365 UBS Securities Australie Ltd Buy 66,359 7,365 UBS Securities Australie Ltd Buy 2,05,533 2,000 UBS Securities Australie Ltd Buy 2,000 3,300 UBS Securities Australie Ltd Buy 7,117 30 UBS Securities Australie Ltd Buy 7,51,25 8,338 UBS Securities Australie Ltd Buy 7,51,26 8,338 UBS Securities Australie Ltd Buy 7,344 1,544 1,286 UBS Securities Australie Ltd Buy 7,344 1,544 1,286 UBS Securities Australie Ltd Buy 7,344 1,444 1,7		UBS Securities Australia Ltd	Buy	164.324	18.238	Ordinary
UBS Securities Australia Ltd Buy 6,359 7,365 UBS Securities Australia Ltd Buy 66,359 7,365 UBS Securities Australia Ltd Buy 205,563 22,815 UBS Securities Australia Ltd Buy 205,563 22,815 UBS Securities Australia Ltd Buy 205,563 2,2815 UBS Securities Australia Ltd Buy 205,363 4,306 UBS Securities Australia Ltd Buy 38,723 4,306 UBS Securities Australia Ltd Buy 1,17 134 UBS Securities Australia Ltd Buy 2,011 5,44 UBS Securities Australia Ltd Buy 1,547 1,84 UBS Securities Australia Ltd Buy 2,018 1,543 UBS Securities Australia Ltd Buy 2,018 1,544 UBS Securities Australia Ltd Buy 2,018 1,544 UBS Securities Australia Ltd Buy 2,018 1,547 UBS Securities Australia Ltd Buy 2,018 1,547 UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Six	3 640	404	Ordinary
UBS Securities Australia Ltd Buy 66,359 7,365 UBS Securities Australia Ltd Buy 66,359 7,365 UBS Securities Australia Ltd Buy 20,369 7,365 UBS Securities Australia Ltd Buy 27,330 3,305 UBS Securities Australia Ltd Buy 27,330 3,005 UBS Securities Australia Ltd Buy 27,330 1,386 1,579 UBS Securities Australia Ltd Buy 75,125 8,386 1,579 UBS Securities Australia Ltd Buy 7,512 8,38 UBS Securities Australia Ltd Buy 7,512 8,38 UBS Securities Australia Ltd Buy 7,512 8,24 UBS Securities Australia Ltd Buy 7,014 1,528 07 UBS Securities Australia Ltd Buy 2,014 2,014	-May-14	I IBS Securities Australia I to	Burn	0	-	1
UBS Securities Australia Ltd Buy 06.359 1,750 UBS Securities Australia Ltd Buy 205,563 22,815 UBS Securities Australia Ltd Buy 205,563 22,815 UBS Securities Australia Ltd Buy 205,563 22,815 UBS Securities Australia Ltd Buy 38,707 4,306 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 7,014 2,244 UBS Securities Australia Ltd Buy 7,014 2,244 UBS Securities Australia Ltd Buy 7,014 2,244 UBS Securities Australia Ltd Buy 7,014 1,725 UBS Securities Australia Ltd Buy 7,014 1,725 UBS Securities Australia Lt		Late citation A societies of 200	, and	D		Cramary
UBS Securities Australia Ltd Buy 205,593 2.8.16 UBS Securities Australia Ltd Buy 27,030 3.000 UBS Securities Australia Ltd Buy 27,030 3.000 UBS Securities Australia Ltd Buy 117 39 UBS Securities Australia Ltd Buy 1,38 154 UBS Securities Australia Ltd Buy 1,38 1,54 UBS Securities Australia Ltd Buy 7,5125 8,338 UBS Securities Australia Ltd Buy 7,5125 8,338 UBS Securities Australia Ltd Buy 1,288 1,544 UBS Securities Australia Ltd Buy 1,528 1,544 UBS Securities Australia Ltd Buy 1,548 1,73 UBS Securities Australia Ltd Buy 1,548 1,73 UBS Securities Australia Ltd Buy 1,548 1,244 UBS Securities Australia Ltd Buy 2,014 2,549 UBS Securities Australia Ltd Buy 2,014 2,549 UBS Securities Australia Ltd Bu	-Ividy-14	Obs securities Australia Lid	euy	655,400	495'/	Ordinary
UBS Securities Australia Ltd Buy 205,563 2.8 B15 UBS Securities Australia Ltd Buy 27,030 30,000 UBS Securities Australia Ltd Buy 117 13 UBS Securities Australia Ltd Buy 1,68.36 1,530 UBS Securities Australia Ltd Buy 1,78,125 8,338 UBS Securities Australia Ltd Buy 1,78,125 8,338 UBS Securities Australia Ltd Buy 1,78,125 8,338 UBS Securities Australia Ltd Buy 1,295 15,743 UBS Securities Australia Ltd Buy 1,647 1,84 UBS Securities Australia Ltd Buy 2,014 2,544 UBS Securities Australia Ltd Buy 2,014 3,24 UBS Securities Australia Ltd Buy 2,918 1,58 UBS Securities Australia Ltd Buy 2,918 1,08 UBS Securities Australia Ltd Buy 2,318 1,08 UBS Securities Australia Ltd Buy 2,318 1,08 UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Buy	54	9	Ordinary
UBS Securities Australia Ltd Buy 37.030 3.000 UBS Securities Australia Ltd Buy 14,505 4.306 UBS Securities Australia Ltd Buy 14,6,535 4.506 UBS Securities Australia Ltd Buy 117 13 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 1,647 1,848 UBS Securities Australia Ltd Buy 3,014 2,75 UBS Securities Australia Ltd Buy 3,04 4,24 UBS Securities Australia Ltd Buy 3,04 1,75 UBS Securities Australia Ltd Buy 3,04 1,75 1,25 UBS Securities Australia Ltd Buy 3,04 1,75 1,25 UBS Securities Australia Ltd Buy 2,918 1,708 1,05 UBS Securities Australia Ltd Buy 2,918 1,05 1,0	-May-14	UBS Securities Australia Ltd	Buy	205,563	22,815	Ordinary
URS Securities Australia Ltd Buy 38,797 4,306 URS Securities Australia Ltd Buy 117 13 URS Securities Australia Ltd Buy 117 13 URS Securities Australia Ltd Buy 75,125 8,338 URS Securities Australia Ltd Buy 75,125 8,338 URS Securities Australia Ltd Buy 75,125 8,338 URS Securities Australia Ltd Buy 42,295 5,549 URS Securities Australia Ltd Buy 42,295 5,549 URS Securities Australia Ltd Buy 42,295 15,44 URS Securities Australia Ltd Buy 1,647 184 URS Securities Australia Ltd Buy 2,018 3,244 URS Securities Australia Ltd Buy 1,1474 1,132 URS Securities Australia Ltd Buy 2,918 3,66 URS Securities Australia Ltd Buy 2,336 1,08 URS Securities Australia Ltd Buy 2,344 1,08 URS Securities Australia Ltd Buy<	J-May-14	UBS Securities Australia Ltd	Buy	27,030	3,000	Ordinary
UBS Securities Australia Ltd Buy 146,836 16,297 UBS Securities Australia Ltd Buy 811 90 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 75,017 15,43 UBS Securities Australia Ltd Buy 49,996 5,549 UBS Securities Australia Ltd Buy 49,996 5,549 UBS Securities Australia Ltd Buy 1,647 1,844 UBS Securities Australia Ltd Buy 3,94 44 UBS Securities Australia Ltd Buy 1,548 1,78 UBS Securities Australia Ltd Buy 2,918 3,26 UBS Securities Australia Ltd Buy 1,1474 1,28 UBS Securities Australia Ltd Buy 2,918 2,014 1,28 UBS Securities Australia Ltd Buy 2,918 1,08 UBS Securities Australia Ltd Buy 2,918 1,08 UBS Securities Australia L	-May-14	UBS Securities Australia Ltd	Buv	38.797	4 306	Ordinary
UBS Securities Australia Ltd Buy H17 19 UBS Securities Australia Ltd Buy 117 13 UBS Securities Australia Ltd Buy 75,128 8,38 UBS Securities Australia Ltd Buy 75,128 8,538 UBS Securities Australia Ltd Buy 75,128 15,43 UBS Securities Australia Ltd Buy 1,647 184 UBS Securities Australia Ltd Buy 1,647 184 UBS Securities Australia Ltd Buy 2,014 2,244 UBS Securities Australia Ltd Buy 2,014 2,245 UBS Securities Australia Ltd Buy 2,018 3,01 UBS Securities Australia Ltd Buy 7,34 1,38 UBS Securities Australia Ltd Buy 7,34 1,08 UBS Securities Australia Ltd Buy <td< td=""><td>May-14</td><td>I IBS Securities Australia I to</td><td>a a</td><td>10000</td><td>500.31</td><td></td></td<>	May-14	I IBS Securities Australia I to	a a	10000	500.31	
UBS Securities Australia Ltd Buy 117 19 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 75,125 8,338 UBS Securities Australia Ltd Buy 49,996 5,549 UBS Securities Australia Ltd Buy 49,996 5,549 UBS Securities Australia Ltd Buy 1,647 1844 UBS Securities Australia Ltd Buy 38,238 4,244 UBS Securities Australia Ltd Buy 3,014 2,74 UBS Securities Australia Ltd Buy 3,014 1,73 UBS Securities Australia Ltd Buy 2,918 3,26 UBS Securities Australia Ltd Buy 2,918 3,26 UBS Securities Australia Ltd Buy 2,918 3,26 UBS Securities Australia Ltd Buy 2,918 1,08 UBS Securities Australia Ltd Buy 2,918 1,08 UBS Securities Australia Ltd Buy 2,918 1,08 UBS Securities Australia Ltd Buy <td>1420.14</td> <td>1100 Countries Australia 14-d</td> <td>Son o</td> <td>00000</td> <td>10,237</td> <td>Cidinaly</td>	1420.14	1100 Countries Australia 14-d	Son o	00000	10,237	Cidinaly
UBS Securities Australia Ltd Buy 75,128 8.38 UBS Securities Australia Ltd Buy 75,128 8.38 UBS Securities Australia Ltd Buy 4,996 5,59 UBS Securities Australia Ltd Buy 4,996 5,44 UBS Securities Australia Ltd Buy 4,244 2,244 UBS Securities Australia Ltd Buy 2,011 2,57 UBS Securities Australia Ltd Buy 2,014 2,57 UBS Securities Australia Ltd Buy 2,014 2,57 UBS Securities Australia Ltd Buy 2,018 1,73 UBS Securities Australia Ltd Buy 7,74 82 UBS Securities Australia Ltd Buy 7,74 82 UBS Securities Australia Ltd Buy 4,708 2,918 3,75 UBS Securities Australia Ltd Buy 2,318 1,78 1,08 UBS Securities Australia Ltd Buy 2,318 1,07 UBS Securities Australia Ltd Buy 2,318 1,71 UBS Securities A	-ividy- 14	ODS SECULINES AUSTRAIN LIU	euy	118	OS	Ordinary
UBS Securities Australia Ltd Buy 7,7125 8,338 UBS Securities Australia Ltd Buy 1,388 1,544 UBS Securities Australia Ltd Buy 4,996 5,549 UBS Securities Australia Ltd Buy 142,295 15,793 UBS Securities Australia Ltd Buy 1,647 1,64 UBS Securities Australia Ltd Buy 38,238 4,244 UBS Securities Australia Ltd Buy 2,014 2,24 UBS Securities Australia Ltd Buy 3,94 4,1 UBS Securities Australia Ltd Buy 7,34 82 UBS Securities Australia Ltd Buy 2,918 3,26 UBS Securities Australia Ltd Buy 2,918 3,26 UBS Securities Australia Ltd Buy 2,918 1,028 UBS Securities Australia Ltd Buy 2,326 1,028 UBS Securities Australia Ltd Buy 2,336 1,028 UBS Securities Australia Ltd Buy 2,336 1,028 UBS Securities Australia Ltd Buy	- 1	UBS Securities Australia Ltd	Buy	117	13	Ordinary
UBS Securities Australia Ltd Buy 1,388 154 UBS Securities Australia Ltd Buy 5,001 555 UBS Securities Australia Ltd Buy 49,996 5,494 UBS Securities Australia Ltd Buy 142,295 15,793 UBS Securities Australia Ltd Buy 38,238 4,244 UBS Securities Australia Ltd Buy 5,014 225 UBS Securities Australia Ltd Buy 5,178 42 UBS Securities Australia Ltd Buy 1,548 1,73 UBS Securities Australia Ltd Buy 1,548 1,73 UBS Securities Australia Ltd Buy 1,744 1,222 UBS Securities Australia Ltd Buy 4,708 5,78 UBS Securities Australia Ltd Buy 2,318 1,08 UBS Securities Australia Ltd Buy		UBS Securities Australia Ltd	Buy	75,125	8,338	Ordinary
UBS Securities Australia Ltd Buy 4,996 5,549 UBS Securities Australia Ltd Buy 4,299 5,449 UBS Securities Australia Ltd Buy 1,42,293 1,544 UBS Securities Australia Ltd Buy 2,014 2,54 UBS Securities Australia Ltd Buy 2,014 2,52 UBS Securities Australia Ltd Buy 2,014 2,52 UBS Securities Australia Ltd Buy 2,014 2,52 UBS Securities Australia Ltd Buy 2,918 3,2 UBS Securities Australia Ltd Buy 1,1474 1,282 UBS Securities Australia Ltd Buy 4,708 2,018 UBS Securities Australia Ltd Buy 2,336 1,08 UBS Securities Australia Ltd Buy<		UBS Securities Australia Ltd	Buy	1,388	154	Ordinary
UBS Securities Australia Ltd Buy 43,996 5,549 UBS Securities Australia Ltd Buy 142,295 15,793 UBS Securities Australia Ltd Buy 1,647 184 UBS Securities Australia Ltd Buy 3,94 4,24 UBS Securities Australia Ltd Buy 3,94 4,24 UBS Securities Australia Ltd Buy 7,94 82 UBS Securities Australia Ltd Buy 7,94 82 UBS Securities Australia Ltd Buy 7,94 82 UBS Securities Australia Ltd Buy 7,91 7,76 UBS Securities Australia Ltd Buy 2,918 7,76 UBS Securities Australia Ltd Buy 2,918 7,76 UBS Securities Australia Ltd Buy 2,918 1,08 UBS Securities Australia Ltd Buy 2,33 1,08 UBS Securities Australia Ltd Buy 2,38 267 UBS Securities Australia Ltd Buy 2,38 1,07 UBS Securities Australia Ltd Buy 2,	l	UBS Securities Australia I to	RIN	5,001	252	Ordinary
UBS Securities Australia Ltd Buy 142,295 15,733 UBS Securities Australia Ltd Buy 1647 1548 UBS Securities Australia Ltd Buy 2,014 2,25 UBS Securities Australia Ltd Buy 2,014 2,75 UBS Securities Australia Ltd Buy 2,014 2,75 UBS Securities Australia Ltd Buy 7,734 82 UBS Securities Australia Ltd Buy 1,755 1,282 UBS Securities Australia Ltd Buy 1,1474 1,282 UBS Securities Australia Ltd Buy 1,714 1,282 UBS Securities Australia Ltd Buy 4,708 2,518 UBS Securities Australia Ltd Buy 2,318 1,088 UBS Securities Australia Ltd Buy 2,336 2,61 UBS Securities Australia Ltd Buy 2,336 2,61 UBS Securities Australia Ltd Buy 2,336 1,07 UBS Securities Australia Ltd Buy 2,336 1,07 UBS Securities Australia Ltd Buy </td <td>ı</td> <td>I IBS Securities Australia I td</td> <td>Burk</td> <td>19 006</td> <td>000</td> <td>Ciding,</td>	ı	I IBS Securities Australia I td	Burk	19 006	000	Ciding,
UBS Securities Australia Ltd Buy 14,243-9 15,444 UBS Securities Australia Ltd Buy 1,647 1844 UBS Securities Australia Ltd Buy 2,014 225 UBS Securities Australia Ltd Buy 394 47 UBS Securities Australia Ltd Buy 734 173 UBS Securities Australia Ltd Buy 734 173 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,918 1,088 UBS Securities Australia Ltd Buy 2,336 1,09 UBS Securities Australia Ltd Buy 2,318 1,09 UBS Securities Australia Ltd Buy 2,318 1,09 UBS Securities Australia Ltd Buy 2,318 1,09 UBS Securities Australia Ltd Buy	NAN 10	- IDC Countries Australia Las	600	066,64	2,743	Cicilary
UBS Securities Australia Ltd Buy 38,238 4,244 UBS Securities Australia Ltd Buy 1647 225 UBS Securities Australia Ltd Buy 2,014 225 UBS Securities Australia Ltd Buy 510 57 UBS Securities Australia Ltd Buy 1,548 173 UBS Securities Australia Ltd Buy 1,548 173 UBS Securities Australia Ltd Buy 1,548 1,73 UBS Securities Australia Ltd Buy 1,735 1,282 UBS Securities Australia Ltd Buy 4,706 5,76 UBS Securities Australia Ltd Buy 2,336 261 UBS Securities Australia Ltd Buy 2,336 261 UBS Securities Australia Ltd Buy 2,336 261 UBS Securities Australia Ltd Buy 2,336 1,08 UBS Securities Australia Ltd Buy 2,336 1,08 UBS Securities Australia Ltd Buy 5,37 60 UBS Securities Australia Ltd Buy 5,3	things.	COO Securities Australia Liu	Buy	142,235	15,73	Ordinary
UBS Securities Australia Ltd Buy 1647 184 UBS Securities Australia Ltd Buy 2,014 44 UBS Securities Australia Ltd Buy 394 44 UBS Securities Australia Ltd Buy 1,548 173 UBS Securities Australia Ltd Buy 734 82 UBS Securities Australia Ltd Buy 734 82 UBS Securities Australia Ltd Buy 1,155 1,28 UBS Securities Australia Ltd Buy 4,708 55 UBS Securities Australia Ltd Buy 2,918 3.6 UBS Securities Australia Ltd Buy 2,78 6.0 UBS Securities Australia Ltd Buy 2,78 6.0 UBS Securities Australia Ltd Buy 2,38 1,08 UBS Securities Australia Ltd Buy 2,38 1,07 UBS Securities Australia Ltd Buy 5,37 6.0 UBS Securities Australia Ltd Buy 5,38 1,07 UBS Securities Australia Ltd Buy 5,38 <t< td=""><td>-May-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>38,238</td><td>4,244</td><td>Ordinary</td></t<>	-May-14	UBS Securities Australia Ltd	Buy	38,238	4,244	Ordinary
UBS Securities Australia Ltd Buy 2,014 225 UBS Securities Australia Ltd Buy 510 57 UBS Securities Australia Ltd Buy 1,548 173 UBS Securities Australia Ltd Buy 1,548 173 UBS Securities Australia Ltd Buy 2,59 33 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,386 261 UBS Securities Australia Ltd Buy 2,386 261 UBS Securities Australia Ltd Buy 2,386 261 UBS Securities Australia Ltd Buy 2,37 60 UBS Securities Australia Ltd Buy 537 6	⊱May-14	UBS Securities Australia Ltd	Buy	1,647	184	Ordinary
UBS Securities Australia Ltd Buy 394 44 UBS Securities Australia Ltd Buy 510 57 UBS Securities Australia Ltd Buy 734 82 UBS Securities Australia Ltd Buy 734 82 UBS Securities Australia Ltd Buy 7,152 1,282 UBS Securities Australia Ltd Buy 2,918 356 UBS Securities Australia Ltd Buy 4,708 2,56 UBS Securities Australia Ltd Buy 2,336 2,61 UBS Securities Australia Ltd Buy 2,336 2,61 UBS Securities Australia Ltd Buy 2,336 1,088 UBS Securities Australia Ltd Buy 2,336 1,08 UBS Securities Australia Ltd Buy 2,336 1,07 UBS Securities Australia Ltd Buy 2,336 1,07 UBS Securities Australia Ltd Buy 2,387 60 UBS Securities Australia Ltd Buy 2,387 1,07 UBS Securities Australia Ltd Buy 2,387 <td>I-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>2.014</td> <td>775</td> <td>Ordinary</td>	I-May-14	UBS Securities Australia Ltd	Buv	2.014	775	Ordinary
UBS Securities Australia Ltd Buy 510 57 UBS Securities Australia Ltd Buy 1,548 173 UBS Securities Australia Ltd Buy 7,94 82 UBS Securities Australia Ltd Buy 1,175 129 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,738 2,61 UBS Securities Australia Ltd Buy 2,738 2,61 UBS Securities Australia Ltd Buy 5,73 60 UBS Securities Australia Ltd Buy 5,37 60 UBS Securities Australia Ltd Buy 6,37 7 UBS Securities Australia Ltd Buy 6,38 1,07 UBS Securities Australia Ltd Buy 6,38		UBS Securities Australia Ltd	202	304	N N	Ording
UBS Securities Australia Ltd Buy 1.54B 137 UBS Securities Australia Ltd Buy 1.54B 137 UBS Securities Australia Ltd Buy 1.152 129 UBS Securities Australia Ltd Buy 1.1474 1.282 UBS Securities Australia Ltd Buy 4.70B 5.01B UBS Securities Australia Ltd Buy 4.70B 5.01 UBS Securities Australia Ltd Buy 7.33B 1.08B UBS Securities Australia Ltd Buy 5.37 60 UBS Securities Australia Ltd Buy 6.05 90 UBS Securities Australia Ltd Buy 6.05	L	IBC Counties Australia 144	652	100		Ordinary
UBS Securities Australia Ltd Buy 15.48 173 UBS Securities Australia Ltd Buy 734 813 UBS Securities Australia Ltd Buy 734 82 UBS Securities Australia Ltd Buy 1,155 128 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,738 1,088 UBS Securities Australia Ltd Buy 2,738 1,088 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 538 107 UBS Securities Australia Ltd Buy 60 90 UBS Securities Australia Ltd Buy 1,781 199	- 1	UBS Securities Australia Ltd	Buy	510	27	Ordinary
UBS Securities Australia Ltd Buy 724 82 UBS Securities Australia Ltd Buy 734 82 UBS Securities Australia Ltd Buy 1,155 129 UBS Securities Australia Ltd Buy 4,708 256 UBS Securities Australia Ltd Buy 4,708 256 UBS Securities Australia Ltd Buy 2,336 261 UBS Securities Australia Ltd Buy 2,336 261 UBS Securities Australia Ltd Buy 2,336 1,088 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 1,065 1,0 UBS Securities Australia Ltd Buy 2,7 3 UBS Securities Australia Ltd Buy 60 90 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 63 90 UBS Securities Australia Ltd Buy 63 90 UBS Securities Australia Ltd Buy 63 90 </td <td>- 1</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>1,548</td> <td>173</td> <td>Ordinary</td>	- 1	UBS Securities Australia Ltd	Buy	1,548	173	Ordinary
UBS Securities Australia Ltd Buy 255 33 UBS Securities Australia Ltd Buy 1,155 129 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 2,336 1,088 UBS Securities Australia Ltd Buy 2,336 1,088 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 538 107 UBS Securities Australia Ltd Buy 538 107 UBS Securities Australia Ltd Buy 538 100		UBS Securities Australia Ltd	Buy	734	82	Ordinary
UBS Securities Australia Ltd Buy 1,155 1282 UBS Securities Australia Ltd Buy 1,1474 1,282 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 4,708 526 UBS Securities Australia Ltd Buy 2,336 261 UBS Securities Australia Ltd Buy 2,37 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 1,056 107 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 63 90 UBS Securities Australia Ltd Buy 63 90 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,387 267 UBS Securities Australia Ltd Buy 1,781 199	-Mav-14	UBS Securities Australia Ltd	Bliv	295	33	Ordinary
UBS Securities Australia Ltd Buy 11,473 1,822 UBS Securities Australia Ltd Buy 2,918 326 UBS Securities Australia Ltd Buy 4,708 526 UBS Securities Australia Ltd Buy 2,336 2,518 UBS Securities Australia Ltd Buy 2,736 1,088 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 538 107 UBS Securities Australia Ltd Buy 896 90 UBS Securities Australia Ltd Buy 896 90 UBS Securities Australia Ltd Buy 1,781 199 </td <td>-May-14</td> <td>HRS Securities Australia Ltd</td> <td>B</td> <td>1155</td> <td>000</td> <td>(in in in</td>	-May-14	HRS Securities Australia Ltd	B	1155	000	(in in in
UBS Securities Australia Ltd Buy 11,474 1,282 UBS Securities Australia Ltd Buy 1,708 2,58 UBS Securities Australia Ltd Buy 4,706 2,58 UBS Securities Australia Ltd Buy 2,386 261 UBS Securities Australia Ltd Buy 2,73 1,088 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 1,710 1,24 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 8,36 1,781 1,99 UBS Securities Australia Ltd Buy 2,387 2,67 UBS Securities Australia Ltd Buy 2,387 2,67 UBS Securities Australia Ltd Buy	-lada-l-	OBS Securities Australia LIU	and	551,1	671	Ordinary
UBS Securities Australia Ltd Buy 2.918 326 UBS Securities Australia Ltd Buy 4,708 526 UBS Securities Australia Ltd Buy 2,336 2,51 UBS Securities Australia Ltd Buy 2,336 1,088 UBS Securities Australia Ltd Buy 5,37 60 UBS Securities Australia Ltd Buy 5,38 1,07 UBS Securities Australia Ltd Buy 6,38 1,07 UBS Securities Australia Ltd Buy 6,38 1,07 UBS Securities Australia Ltd Buy 8,56 9,0 UBS Securities Australia Ltd Buy 8,58 1,07 UBS Securities Australia Ltd Buy 2,387 2,67 UBS Securities Australia Ltd Buy 3,67	-May-14	UBS Securities Australia Ltd	Buy	11,474	1,282	Ordinary
UBS Securities Australia Ltd Buy 4.708 526 UBS Securities Australia Ltd Buy 2,336 261 UBS Securities Australia Ltd Buy 9,738 1,088 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 57 3 UBS Securities Australia Ltd Buy 1,110 1,24 UBS Securities Australia Ltd Buy 2,387 1,07 UBS Securities Australia Ltd Buy 2,387 1,00 UBS Securities Australia Ltd Buy 2,220 2,48 UBS Securities Australia Ltd Buy 3,029	-May-14	UBS Securities Australia Ltd	Виу	2,918	326	Ordinary
UBS Securities Australia Ltd Buy 2,336 261 UBS Securities Australia Ltd Buy 9,738 1,088 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 66 90 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 63 90 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 3,097 348 UBS Securities Australia Ltd Buy 3,79 344	-May-14	UBS Securities Australia Ltd	Buy	4,708	276	Ordinary
UBS Securities Australia Ltd Buy 9,738 1,088 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 7 7 UBS Securities Australia Ltd Buy 558 107 UBS Securities Australia Ltd Buy 67 124 UBS Securities Australia Ltd Buy 68 90 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 806 100 UBS Securities Australia Ltd Buy 806 100 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 344 UBS Securities Australia Ltd Buy 3,079 44	1-May-14	UBS Securities Australia Ltd	Buy	2,336	261	Ordinary
UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 1,065 119 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 63 107 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 8,387 2,67 UBS Securities Australia Ltd Buy 8,387 107 UBS Securities Australia Ltd Buy 8,387 109 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 248 UBS Securities Australia Ltd Buy 3,079 248 UBS Securities Australia Ltd Buy 3,079 248	-Mav-14	UBS Securities Australia Ltd	Bit	9 738	1088	Ordinan
UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 72 3 UBS Securities Australia Ltd Buy 928 107 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 826 90 UBS Securities Australia Ltd Buy 836 90 UBS Securities Australia Ltd Buy 836 107 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,287 109 UBS Securities Australia Ltd Buy 3,079 344	-May-14	I IBS Securities Australia 14d	Britis	76	200	Ordina.
UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 1,065 119 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 63 107 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 2,387 267 UBS Securities Australia Ltd Buy 895 100 UBS Securities Australia Ltd Buy 2,387 100 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,720 248 UBS Securities Australia Ltd Buy 3,720 248 UBS Securities Australia Ltd Buy 3,039 344 UBS Securities Australia Ltd Buy 3,039 344	14 m	The Countries Australia Ltd	had	77	2 (Ordinary
UBS Securities Australia Ltd Buy 537 60 UBS Securities Australia Ltd Buy 1,065 19 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 958 107 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 63 90 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 805 90 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 3,220 248 UBS Securities Australia Ltd Buy 3,09 34 UBS Securities Australia Ltd Buy 3,09 34 UBS Securities Australia Ltd Buy 3,09 34	-lvidy-14	Obs securities Australia Ltd	Buy	53/	06	Ordinary
UBS Securities Australia Ltd Buy 1,065 119 UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 958 107 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 2,387 267 UBS Securities Australia Ltd Buy 2,387 100 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 348 UBS Securities Australia Ltd Buy 3,079 348 UBS Securities Australia Ltd Buy 3,079 348 UBS Securities Australia Ltd Buy 814 91	-May-14	UBS Securities Australia Ltd	Buy	537	90	Ordinary
UBS Securities Australia Ltd Buy 27 3 UBS Securities Australia Ltd Buy 558 107 UBS Securities Australia Ltd Buy 1,110 1,74 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 2,387 100 UBS Securities Australia Ltd Buy 895 100 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,020 248 UBS Securities Australia Ltd Buy 3,020 248 UBS Securities Australia Ltd Buy 814 91	⊱May-14	UBS Securities Australia Ltd	Buy	1,065	119	Ordinary
UBS Securities Australia Ltd Buy 958 107 UBS Securities Australia Ltd Buy 1,110 124 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 865 100 UBS Securities Australia Ltd Buy 875 100 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 48 UBS Securities Australia Ltd Buy 814 91	-May-14	UBS Securities Australia Ltd	Buy	27	3	Ordinary
UBS Securities Australia Ltd Buy 1,110 124 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 2,387 267 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 348 UBS Securities Australia Ltd Buy 814 91	-May-14	UBS Securities Australia Ltd	Buy	958	107	Ordinary
UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 63 7 UBS Securities Australia Ltd Buy 2,387 267 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 344 UBS Securities Australia Ltd Buy 814 91	-May-14	UBS Securities Australia 11d	Bisy	1 110	124	Colingo
Usbs Securities Australia Ltd Buy Bob 90 UBS Securities Australia Ltd Buy 2,387 267 UBS Securities Australia Ltd Buy 895 100 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 344 UBS Securities Australia Ltd Buy 814 91	,	The Committee Australia Lat	600	011,1	17	Ci Cilitary
UBS Securities Australia Ltd Buy 806 90 UBS Securities Australia Ltd Buy 2,387 267 UBS Securities Australia Ltd Buy 895 100 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 3,720 248 UBS Securities Australia Ltd Buy 3,079 344 UBS Securities Australia Ltd Buy 814 91	-INIGN-14	OBS Securities Australia Ltd	pris	69	`	Orginary
UBS Securities Australia Ltd Buy 2.387 267 UBS Securities Australia Ltd Buy 895 100 UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 344 UBS Securities Australia Ltd Buy 814 91	-May-14	UBS Securities Australia Ltd	Buy	908	90	Ordinary
UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 344 UBS Securities Australia Ltd Buy 814 91	-May-14	UBS Securities Australia Ltd	Buy	2,387	267	Ordinary
UBS Securities Australia Ltd Buy 1,781 199 UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,09 344 UBS Securities Australia Ltd Buy 814 91	-May-14	UBS Securities Australia Ltd	Buv	895	100	Ordinary
UBS Securities Australia Ltd Buy 2,220 248 UBS Securities Australia Ltd Buy 3,079 344 UBS Securities Australia Ltd Buy 814 91	-May-14	I IBS Securities Australia I to	Birk	1 781	001	Ordinan,
UBS Securities Australia Ltd Buy 3,079 3.44 UBS Securities Australia Ltd Buy 814 91	-Mav-14	JRS Securities Australia Ltd	Bix	2 220	2/8	Ordinary
UBS Securities Australia Ltd Buy 814 91	May 14	1 DC Cooperation Australia Ltd		0100	24.7	(Interior
UBS Securities Australia Ltd Buy 814 91	-lviay-	ספס ספרתו ונוכם שמפוז מום ורוס	Ang	1 Z / N.C		
	-May-14				F.	Orginary

10 110 11	II IDC Consisting Australia 144		1 823	100	- Pro-C
19-May-14	UBS Securities Australia Ltd	auy B.n.	1,823	107	Ordinary
19-May-14	UDS Securities Australia Ltd	Buy	1 487	900	Ordinary
19-Ividy-14	UBS Securities Australia Lid	anà b:::	1,48/	40 5	Ordinary
19-May-14	Ubs Securities Australia Ltd	gay	/01/1	771	Ordinary
19-May-14		Buy	2,464	272	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,902	320	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	9,070	1,000	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	11,972	1,320	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	3,153	348	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,703	298	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	4.898	540	Ordinary
19-May-14	I IBS Securities Australia I to	Rin	11 492	1 267	Ordinan
19-May-14	I IRS Societies Australia 14d	2 3	1751	102,1	Ordinary
10 May 14	190 Con which America 14d	pay.	771 C	080	Ordinary
10 th 14	Coo Securities Australia Liu	boy	//1/7	047	Ordinary
- 1	UBS Securities Australia Ltd	Buy	1,624	1/9	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	10,893	1,201	Ordinary
- 1	UBS Securities Australia Ltd	Buy	526	28	Ordinary
	UBS Securities Australia Ltd	Buy	562	62	Ordinary
	UBS Securities Australia Ltd	Buy	526	85	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	36	7	Ordinary
ı	UBS Securities Australia Ltd	Buy	8,689	958	Ordinary
1	I IBS Securities Australia I to	A. O.	10.703	1 190	Ordinan
10-6434-14	1 IBC Constitute Australia 14d	100	02/2	201,1	y in it
	1 DC Commission Australia Hol	900	1,27,3	200	Cidinary
O May 14	DO SCHOOL ACTION AND AND AND AND AND AND AND AND AND AN	buy P	606,1	017	Ordinary
3-ividy- 14	Upo Securities Australia Lid	any	970/6	400	Ordinary
3-IMIdy-14	Obs securities Australia Ltd	buy	806,6	660'1	Ordinary
19-May-14	UBS Securities Australia Ltd	Ruy	7/7/7	751	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	8,072	880	Ordinary
19-May-14	UBS Securities Australia Ltd	8uy	10,512	1,159	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,620	583	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	2,131	235	Ordinary
	UBS Securities Australia Ltd	Buy	10,267	1,132	Ordinary
	UBS Securities Australia Ltd	Buy	10,267	1.132	Ordinary
19-May-14	UBS Securities Australia Ltd	Buv	986'6	1,101	Ordinary
	UBS Securities Australia (1d	Sirv	3 156	348	Ordinary
9-May-14	UBS Securities Australia I td	Birv	3 301	364	Ordinary
9-Mav-14	I IRS Securities Australia I to	Rin	98	~	Ordinany
	I IBC Contrition Australia 14d	, a	0000	100	Ordingo
10 May 14	billococcupies Australia India	600	000	167	Ordinary
1	1 DE Constitut Australia 1+4		500	OF S	Oldinary Oldinary
1	COO SECULIVES MUSICALITY	an)	1110	8 8	Cidinary
3-Ividy-14	DDS Securities Australia Lid	Ang	8,045	/88	Ordinary
- 1	UBS Securities Australia Ltd	Buy	/97,01	1,132	Ordinary
ı	UBS Securities Australia Ltd	Buy	2,222	245	Ordinary
	UBS Securities Australia Ltd	Buy	10,557	1,164	Ordinary
	UBS Securities Australia Ltd	Buy	10,557	1,164	Ordinary
	UBS Securities Australia Ltd	Buy	10,276	1.133	Ordinary
19-Mav-14	UBS Securities Australia Ltd	Buv	7 383	814	Ordinary
9-May-14	I IRS Securities Australia I to	Rin	2 890	310	Ordinan
10 May 14	TIDE Contribute Australia 14d	9.13	2,030	100	Oldinary
3-(vlay- ; 4	מבר כן ביייי אייייי איייייייייייייייייייייייי	bou.	510,01	5 5	Ordinary
9-iviay-14	Ubs securities Australia Ltd	Buy	169'/	848	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	2,449	270	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,061	117	Ordinary
9-May-14	UBS Securities Australia Ltd	8uy	6/0/6	1,00,1	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	1,841	503	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	6.340	669	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buv	3.465	382	Ordinary
19-May-14	LIRS Securities Australia I td	Ric	2 195	242	Ordinary
	LIRA Countries Australia I td	Buy	1 100	121	Ording
O hand 14	OCCUPATION Assistant Live	, ind	2001	756	Ordinary
ı	COO COOLINES AUSTRALIA EN	bus,	6/9/2	/10	Cidinary
1	UBS Securities Australia Ltd	, and	70677	320	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	10,666	1,176	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	10,811	1,192	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	11,084	1,222	Ordinary
19-May-14	UBS Securities Australia Ltd	Buy	11,084	1,222	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	10,947	1.207	Ordinary
9-Mav-14	UBS Securities Australia I td	Riv	10 313	1137	Ordinan
	TIDE Commission Assemble 14-4	100	C1C(C)	100	O'G' I'I
9-May-14	UBS Securities Australia Ltd	Buy	10,313	1,13/	Ordtnary

1				5	Connany
May-14	UBS Securities Australia Ltd	Buy	Sec.	2 8	Ordina
20-May-14	UBS Securities Australia Ltd	Buv	760	84	Ordinary
20-May-14	UBS Securities Australia 11d	Bliv	698	200	Ordings.
20-May-14	I IRS Socienties Australia 11d	A STATE OF THE STA	200	000	Oldinary
1	Laboration Association and	600	626,4	30	Galhary
20 14	Day Significant Common	, and	200,44	4,975	Ordinary
20-Iviay-14	COS Securities Australia Liti	houy	44,999	4,9/5	Ordinary
IVIdy-14	Obs Securities Australia Ltd	gnà	44,999	4,9/5	Ordinary
0-May-+4	UBS Securities Australia Ltd	Buy	42,285	4,675	Ordinary
20-May-14	UBS Securities Australia Ltd	Buy	42,285	4,675	Ordinary
20-May-14	UBS Securities Australia Ltd	Buy	42,285	4,675	Ordinary
0-May-14	UBS Securities Australia Ltd	Buy	42.285	4.675	Ordinary
Mav-14	UBS Securities Australia Ltd	Suv	47 785	4 675	Ordinary
20-May-14	1 BS Securities Australia 11d	30.00	100 07	2,012	
	Lat allegation of the party of the	000	567.24	6/0/4	Cranary
-1-1	COS Seculties Australia Liu	Ána	35,198	7007	Ordinary
U-May-14	UBS Securities Australia Ltd	Buy	38,912	4,302	Ordinary
0-May-14	UBS Securities Australia Ltd	Buy	38,912	4,302	Ordinary
0-May-14	UBS Securities Australia Ltd	Buy	38,912	4,302	Ordinary
20-May-14	UBS Securities Australia Ltd	Buy	38.912	4.302	Ordinary
May-14	UBS Securities Australia Ltd	Buy	38 912	4 302	Ordinary
0-May-14	LIBS Securities Australia Ltd	× 20	EVO 88	300 1	Carding
20-May-14	I IBS Securities Australia 144	6	802.3	7,200	O Colonial y
1	Day of the state o	, and	06/16	40	Ordinary
vidy- 14	Obs Securities Australia Ltd	Buy	4,135	457	Ordinary
U-May-14	UBS Securities Australia Ltd	Buy	6,127	677	Ordinary
May-14	UBS Securities Australia Ltd	Buy	851	94	Ordinary
0-May-14	UBS Securities Australia Ltd	Buy	2,352	260	Ordinary
0-May-14	UBS Securities Australia Ltd	Buy	909	67	Ordinary
0-May-14	UBS Securities Australia 11d	Brix	14.489	1 601	Ordinan
O-May-14	I IBC Countries Australia 14d	77.0	E1C 8	100,1	o delinary
	The control of the state of the	pag.	4,2(1	400	Ordinary
0-INIdy-14	UDS Securities Australia Ltd	auy	2,751	304	Ordinary
May-14	UBS Securities Australia Ltd	Buy	10,036	1,109	Ordinary
May-14	UBS Securities Australia Ltd	Buy	17,620	1,947	Ordinary
20-May-14	UBS Securities Australia Ltd	Buy	20,073	2,218	Ordinary
20-May-14	LBS Securities Australia Ltd	Biv	7 901	873	Ordinary
1-May-14	I RS AG London Branch	9	561.01	(200.3/	dia i
21 May 14	1 DC AC London Drongth		211/01	105.20	, di iai ,
viey .	ייייי אייייייייייייייייייייייייייייייי	áng.	46,016	5,235	Ordinary
-Ividy-14	ODS SECURITIES AUSTRAINA L'IO	Buy	77,468	7/4/7	Ordinary
1-May-14	UBS Securities Australia Ltd	Seli	300,533	(33,030)	Ordinary
May-14	UBS Securities Australia Ltd	Buy	90,500	10,000	Ordinary
21-May-14	UBS Securities Australia Ltd	Buv	368.352	40.488	Ordinary
1-Adam-14	I IDC Cocurition Australia 14d	100	200,020	100 000	
1	COO SECULINES AUSTRAINS LIU	Jan C	3/0,//4	(40,097)	Ordinary
21-May-14	UBS Securities Australia Ltd	Buy	67,276	7,400	Ordinary
May-14	UBS Securities Australia Ltd	Sell	9,010	(1,000)	Ordinary
1-May-14	UBS Securities Australia Ltd	Buy	45,410	2,000	Ordinary
1-May-14	UBS Securities Australia Ltd	la)	48 172	/E 20E)	Ordinan
34 May 14	I ID Commission Assessed in the		2(1)01	(000,00)	S III
viay-14	ODS SECURITIES AUSTRAINA LICO	Enter into Derivative C	2	(000,12)	Options
21-May-14	UBS Globaí Asset Management	Buy	109,104	11,976	Ordinary
21-May-14	UBS Global Asset Management	Buy	84,600	9,253	Ordinary
-Mav-14	UBS Securities Australia Ltd	Sell	100.211	(10,964)	Ordinary
May-14	LBS Securities Australia I td	la S	133 252	(17.570)	Ordinan
21-May-14	I IBC Contrition Australia 14d	200	203,000	10,000	00000
1 1		100	11/16	907	Cidillary
CI-IVIGY-14	Obs securities Australia Ltd	Sell	77,470	(3,000)	Ordinary
21-May-14	UBS Securities Australia Ltd	Sell	203,785	(22,296)	Ordinary
1-May-14	UBS Securities Australia Ltd	Sell	62,161	(6,801)	Ordinary
May-14	UBS Securities Australia Ltd	Seil	104 388	(11 421)	Ordinary
1-Mav-14	LIBS Securities Australia Ltd	lay	71,804	(7.856)	Ordinary
21 May 14	to for a second process of the second proces	17.0	1,004	(0.00, 1)	Cruitary
VIdy-14	ODS Securities Australia Liu	Sett	13,820	(7,512)	Ordinary
1-May-14	UBS Securities Australia Ltd	Sell	4,022	(440)	Ordinary
1-May-14	UBS Securities Australia Ltd	Sell	3,345	(39E)	Ordinary
21-May-14	UBS Securities Australia Ltd	Sell	12,805	(1.401)	Ordinary
1 - Veh	I IBC Contrition Australia 14d	lo S	20072	(002.01)	Carling C
ti foin	one securities Australia Liu	100	010'/6	(10,/02)	Ordinary
May-14	UBS Securities Australia Ltd	Sell.	56,631	(6,196)	Ordinary
1-May-14	UBS Securities Australia Ltd	Sell	56,613	(6,194)	Ordinary
21-May-14	UBS Securities Australia Ltd	Sell	1,892	(202)	Ordinary
21 May 14					
M2V-14	LIBS Securities Australia I to	و	22,311	(17 (7)	Ordinary

5	is securities Australia Ltd	Buy	800,401	07,140	Ordinary
5 :	UBS Securities Australia Ltd	Sell	46,000	(2,000)	Ordinary
5 5	UBS Securities Australia Ltd	Enter Into Derivative C	V	(408)	Options
1	195 Securities Australia Ltd	Enter Into Delivative C	N/A	(408)	Options
2 May-14 Oc	30 Committee Australia Ltd	Entel Into Delivative	XX.	(200)	Options
5 5	so securities Australia Ltd	citter into Derivative C	AN CCO	(76)	Options
1	DOS GIODAI MOSEL IMANIAGENERIN	Buy	70,027	986,51	Orginary
31:	ODS GIODAI ASSEL IMATAGETTETIC	buy	018'0/	8,301	Ordinary
7	DDS Securities Australia Ltd	Zell	6,797	(156)	Ordinary
3	s securities Australia Ltd	ne.	8,131	(6/8)	Ordinary
4	is Securities Australia Ltd	Sell	7,992	(864)	Ordinary
픠	UBS Securities Australia Ltd	Sell	237,790	(25,707)	Ordinary
5	JBS Securities Australia Ltd	Sell	11,100	(1,200)	Ordinary
n	UBS Securities Australia Ltd	Sell	23,791	(2.572)	Ordinary
5	JBS Securities Australia Ltd	Sell	9,703	(1,049)	Ordinary
15	JBS Securities Australia I td	les	462 500	(50,000)	Ordinary
1	Cocumition Australia 14d	130	000,200	(000,00)	Cidillary
1	DOS Securities Australia Liu	llac llac	000	(56)	Orginary
5	S Securities Australia Ltd	Nell Nell	/03	(76)	Ordinary
5	JBS Securities Australia Ltd	Sell	6,928	(246)	Ordinary
≝	S Securities Australia Ltd	Sell	78,542	(8,491)	Ordinary
3	JBS Securities Australia Ltd	Sell	70,060	(7.574)	Ordinary
2	JBS Securities Australia Ltd	روا	51 171	(5,532)	Ordinan
ı	Constitute Australia 14d	= 0		(200'5)	9
1	S Securities Australia Ltd	liac i	9,500	(/70/1)	Ordinary
2-May-14 UE	UBS Securities Australia Ltd	Sell	2,618	(283)	Ordinary
	S Securities Australia Ltd	Sell	64,165	(6,933)	Ordinary
	JBS Securities Australia Ltd	Seli	92.550	(10.000)	Ordinary
Ð	JBS Securities Australia I td	les	138 825	(15,000)	Ordinan
ľ	IBS Societies Australia 14d	Coll	100 001	(000)	1
1	IDC Countries Australia tod		20,000	(000/51)	O G G I al y
	S Securities Australia Ctu	iiac	n !	(i)	Ordinary
- 1	S Securities Australia Ltd	Sell	46	(5)	Ordinary
2-May-14 UE	UBS Securities Australia Ltd	Sell	46	(2)	Ordinary
	S Securities Australia Ltd	Sell	19	(2)	Ordinary
	UBS Securities Australia Ltd	Sell	28	(3)	Ordinary
_	S Securities Australia Ltd	Sell	65	(2)	Ordinary
22-May-14 UB	UBS Securities Australia Ltd	Sell	6.941	(750)	Ordinary
Г	JBS Securities Australia Ltd	Sell	6.114	(199)	Ordinary
۲	IRS Sequitives Australia 14d	lay.	300	(42)	, adjoor
O May 14	DO Commission Assessed to the	120	000	(74)	Couriery
۳	Convition Australia Ltd	100	000	(0)	Oderaly
1	DDS Securities Australia Ltd	llac	8/7	(30)	Ordinary
22-May-14 UB	UBS Securities Australia Ltd	Sell	389	(42)	Ordinary
	UBS Securities Australia Ltd	Sell	278	(30)	Ordinary
22-May-14 UB	S Securities Australia Ltd	Sell	141,960	(15,347)	Ordinary
1	UBS Securities Australia Ltd	Self	o	ε	Ordinary
t	JBS Securities Australia Ltd	la S	3 515	(380)	(in in in
۲	Consister Australia 14d	100	0207	(200)	9
Ť	Obs securities Australia Lid	nac .	4,0/0	(440)	Ordinary
7	UBS Securities Australia Ltd	Sell	4,625	(200)	Ordinary
8	UBS Securities Australia Ltd	Sell	7,992	(864)	Ordinary
	S Securities Australia Ltd	Sell	4,625	(200)	Ordinary
2-May-14 UB	UBS Securities Australia Ltd	Sell	2.035	(020)	Ordinary
T	IBS Securities Australia 11d	lay.	46	(3)	Ordingo
19	DO Counting Australia Ltd		27.7.4	(000 1)	1000
1	IDC Countries Australia 14d	170	100.7	(200)	Cidiliary
3 15	o permittees mastralia Ltd	nec nec	10/'C	(620)	Ordinary
-May-14 UB	Obside Australia Ltd	Sell	1,0/3	(316)	Ordinary
5	UBS Securities Australia Ltd	Sell	6,225	(673)	Ordinary
3	S Securities Australia Ltd	Sell	37	9	Ordinary
2-May-14 UB	UBS Securities Australia Ltd	Sell	49,321	(5,332)	Ordinary
۳	UBS Securities Australia I td	Self	18 167	(1 96.4)	Ordinary
2	IBC Counties Australia I to	Call	6 225	(623)	in in
1	Securities Australia Liu	2001	677'0	(5/0)	Ordinary
3	UBS Securities Australia Lto	Sell	14,532	(1,571)	Ordinary
3	UBS Securities Australia Ltd	Sell	27,565	(2,980)	Ordinary
<u> </u>	S Securities Australia Ltd	Sell	11,479	(1,241)	Ordinary
3	UBS Securities Australia I td	Sell	15.290	(1 653)	Ordinan
3 2	Commission A make the	====	5,7,5	(000,1)	Oldinal y
7	UBS Securities Australia Ltd	Sell	463	(09)	Ordinary
22-May-14 UB	S Securities Australia Ltd	Sell	4 625		
Г			20,1	(200)	Ordinary
	LIBS Securities Australia (1)	Sell	27,	(200)	Ordinary

Self 128 Self 128 Self 128 Self 1,773 Self 1,726 Self 1,726 Self 1,726 Self 1,726 Self 1,726 Self 1,535 Self 1,5	L	(14) Ordinary				(1 226) Ordinary				(2 352) Ordinary	l	L		L	L	(17) Ordinary	(1,164) Ordinary		(10,129) Ordinary	(1,750) Ordinary	۱	(128) Ordinary		L	945,000 Ordinary	(3) Ordinary	(180) Ordinary	(190) Ordinary		(116) Ordinary				(8) Ordinary		L	(2) Ordinary				(584) Ordinary		L	(1,022) Ordinary		(8,063) Ordinary		1	(20,000) Ordinary		(20,000) Ordinary				L					
	900)	128	128	2.285				6.142	155	21 497	55				6	155					6,224	7,623				27	1,643	550,1	237	1,058	4,961	5,700	6,220	73	188	8.527	18	511	8,801	1,961	5,326	1.343	6,541										27.6	6.851	ļ					
Securities Australia Ltd Securities Australia Ltd	CAB	100	Sell	ll s	9	3	19	3	7	3	Sell	Sell	Sell	Sell	Sell	Sell	Seli	Seli	Sell	Se	lles C	Ř. J		Sell	Buy	Şei	la Sell	100	Sell	Sell	Sell	Sell	Sell	Je Je	3	Sell	lleS	Sell	Seil	Zell Xell	Sell Sell	38	3	Sell	Sell	Sell	#e.	Sell	Sell	Sell	Sell	Sell		Sell	Coll	i en	Buy	Buy Sell	Buy Sell Buy Sall	Buy Sell Buy Sell Sell Suv
	Con wither Australia 14d	Securities Australia I to	S Securities Australia Ltd	5 Securities Australia Ltd	S Securities Australia I td	S Securities Australia I td	S Securities Australia I to	S Securities Australia I td	S Securities Australia I to	5 Securities Australia Ltd	S Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Itd	5 Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Itd	Securities Australia Ltd	UBS Securities Australia Ltd		JBS Securities Australia Ltd	JBS Securities Australia Ltd JBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd

2-May-14	IRC Sportition Ametralia 144	3			
May-14	I IRC Securities Australia 14d		377.	(000)	
2-May-14	UBS Securities Australia Ltd	3	3 635	(303)	Ordinary
22-May-14	I IBS Societies Australia 14d	Coll	1 166	(30.1)	O Gillian
la la	COO Securities Australia Liu	ES .	001,1	(071)	Ordinary
22-May-14	UBS Securities Australia Ltd	Neil Neil	2,146	(232)	Ordinary
7-May-14	UBS Securities Australia Ltd	Ne.	28	(E)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seil	2,165	(234)	Ordinary
May-14	UBS Securities Australia Ltd	Sell	6,124	(995)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	2,710	(533)	Ordinary
22-May-14	UBS Securities Australia Ltd	Iles	14.791	(1.599)	Ordinary
2-May-14	UBS Securities Australia Ltd	195	060.9	(658)	Ordinan
May-14	RS Securities Australia Ltd	3	1 000	(110)	Con in Co
2-May 14	PIEC Consisting Australia Ltd	133	1 130	(011)	Oddinaly
	COS Securities Australia Ltd	- C	671,1	(771)	Orginary
4-INIdy-14	UBS Securities Australia Ltd	Sell	11,5,1	(164)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	1,249	(135)	Ordinary
2-May-14	UBS Securities Australia Ltd	Self	11,452	(1,238)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	1,008	(601)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	4,607	(498)	Ordinary
Mav-14	UBS Securities Australia I td	Sel	940.54	(207 ()	Ordinary
22-May-14	UBS Securities Australia Ltd	lay.	7 350	(255)	Ordingo
2-May-14	I IBS Securities Australia I to	9	1 203	(001)	Ordings.
2-May 14	I IBC Cocurition Australia 14d	100	202,	(05)	Cicinaly
T A STATE OF	COO DECONICION MANAGEMENT	llac l	7,240	(543)	Ordinary
Z-iviay-14	UBS Securities Australia Ltd	Sell	111,1	(120)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	1,463	(158)	Ordinary
May-14	UBS Securities Australia Ltd	Sell	9,260	(1,000)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	19	(2)	Ordinary
2-May-14	UBS Securities Australia Ltd	jeS.	4 454	(481)	Ordinary
NEWS A	I IBC Con mittor Auctralia 14d	100	100	(63)	
May 14	Cop Securities Australia Ltd	190	450	(/4)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	1,093	(318)	Ordinary
May-14	UBS Securities Australia Ltd	Seil	3,222	(348)	Ordinary
May-14	UBS Securities Australia Ltd	Sell	£65'9	(712)	Ordinary
2-May-14	UBS Securities Australia Ltd	Se	2.074	(224)	Ordinary
2-May-14	I IBS Securities Australia 14d	700	01,01	(1 001)	6
1	COO Decarries Australia Coo	ion i	60101	(106'1)	Croinary
Z-Mdy-14	UBS Securities Australia Ltd	Sell	701	î.i.	Ordinary
May-14	UBS Securities Australia Ltd	Seil	3,121	(337)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,019	(110)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	472	(51)	Ordinary
2-May-14	UBS Securities Australia Itd	llas	1018	(110)	Ordinary
2-May-14	I IBC Contrition Australia 14d	le o	2001	(113)	in in
140,14	Late Commission Automatic Late	190	(50')	(211)	Cigliary
2-May-14	OBS Securities Australia Ltd	Self	5,063	(573)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,981	(214)	Ordinary
May-14	UBS Securities Australia Ltd	Sell	1,804	(195)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	10101	(1 092)	Ordinary
2-May-14	I IRS Securities Australia 14d		¥03 0	(900)	Ordinary
Many 1	Life Committee Australia Life	150	*DC'0	(070)	100
2-May-14	UBS Securities Australia Ltd	Seil	9/1	(61.)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	2,331	(252)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	17,649	(1,908)	Ordinary
2-Mav-14	UBS Securities Australia Ltd	Sell	1 499	(167)	Ordinary
2-May-14	I IBS Securities Australia I to	Coll	1 360	(1/0)	in in
	COO Securities Australia Ltd	PC	EOC,1	(041)	Cruirary
2-ividy-14	Obs Securities Australia Lio	Sell	7,532	(989)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	066	(107)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	3,043	(329)	Ordinary
May-14	UBS Securities Australia Ltd	Sell	2.659	(828)	Ordinary
2.May-14	I IRS Securities Australia 14d	Coll	2 36.7	(1967)	i i
talay-1-	ODS Securities Australia Liu)ell	105.6	(100)	Ordinary
Z-Mdy-14	Obs securities Australia Lid	Sell	5,109	(977)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seli	1,138	(123)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	£00′9	(649)	Ordinary
May-14	UBS Securities Australia Ltd	Sell	4,394	(475)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	5.837	(631)	Ordinary
22-May-14	I IDC Countities Australia 14d	Coll	1000	1,000,17	Cramary
May-14	UBS Securities Australia Ltd	Sell	475,8	(1,008)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	16,169	(1,748)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	28	(3)	Ordinary
2-May-14	UBS Sequitities Australia I to	Sell	3 765	(407)	Ordinary
May-14	I IBS Sagurities Australia I to		2 500	(190)	
May	טבוומים שיים ווכחי הביוווים אלו היים	- 24			
			00017	(102)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	18,510	(2,000)	Ordinary

2-May-14	UBS Securities Australia Ltd	9.	981	(106)	Ordinary
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	= 0			
22-Ividy-14	UBS Securities Australia Ltd	198	589	(/4)	Ordinary
197-14	Obsider Australia Lid	II-N	88.18	(an I)	Orginary
22-May-14	UBS Securities Australia Ltd	Sell	589	(/4)	Ordinary
107-14	Upo Securities Australia Liu	Set.	87	(F)	Ordinary
22-Mdy-14	Upg Securities Australia Ltd	Sell	95 5	(q)	Ordinary
22-Mdy-14	Ups Securities Australia Ltd	Nell	8/0,3	(549)	Ordinary
22-ivldy-14	Obsideration Australia Ltd	Sell	95.0	(q)	Ordinary
22-IVIdy-14	UBS Securities Australia Ltd	Sell	8,806	(756)	Ordinary
1dy-14	Ups securities Australia Ltd) Sell	057,6	(000,1)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	2,646	(586)	Ordinary
7-May-14	UBS Securities Australia Ltd	Sell	40,950	(4,427)	Ordinary
2-May-14	UBS securities Australia Ltd	Sell	2,664	(288)	Ordinary
Z-May-14	UBS Securities Australia Ltd	Ser.	3/0	(40)	Ordinary
1ay-14	UBS Securities Australia Ltd	Sell	9,278	(1,003)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	2,208	(263)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	29,831	(3,225)	Ordinary
2-May-14	UBS Securities Australia Ltd	- Xe	80,105	(8,660)	Ordinary
Z-ividy-14	Upo Securities Australia Ltd	Oeii	766'/	(804)	Ordinary
421.14	U.D.C. Commission Attendiby Ltd	120	2003	(020)	Ordinary
22-May-14	UBS Counties Australia Ltd	100	501,2	(233)	Ordinary
1dy-14	OBS Securities Australia Ltd	Jen C	18/3	(679)	Ordinary
2 May 14	UBS Securities Adstraila Ltd	130	760	(04)	Ordinary
2 Mary 14	UBS Securities Australia Ltd	200	886,61	(7,161)	Ordinary
May 14	Tibe counties Australia and	liec liec	3,230	(000,1)	Gallhary
2-May-14	I IRS Societies Australia 11d	Coll	10000	(0,040)	Ordinary
2-May-14	I IRS Societies Australia 14d	P	1712	(105)	ordinal y
2-May-14	I IRO Con without Australia 14d	100	2 452	(201)	Clariary
7-May-14	I IBS Securities Australia 14d	3	750.28	(873)	Ordinany
May-14	I IRS Societies Australia 14d	Cott	500,00	(4,0,3)	Ordinary
Aav-14	LIRS Securities Australia 11d	Tag.	611	(5)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	759	(82)	Ordinary
2-May-14	I IRS Societies Australia 14d	Call	120 001	(20)	Ordinary
2-May-14	1 BS Securities Australia 11d	ll o	0.067	(4,47)	Ordinary Ordinary
2-May 1/	A ISC Contrition Australia Ltd	100	195,000	(000,00)	Coding
14-ve	I IRS Societies Australia I to	3	531	(20,000)	Crolinary
7-May-14	I IRC Societies Australia 14d	liec Soll	/61	(/)	Ordinary
14-14	I IRS Sequities Australia 14d	ll ay	1005	(6)	Ordinan
May 14	I IBC Coourities Australia 14d	200	2,545	(000)	Ordinary
2-May-14	LIBS Secretaries Australia 14d	(ac)	104,401	(4 125)	Ordinary
Many 3.4	Libe constitut Australia Liu	11-5	10,400	(C) (C)	Ordinary
421. 54	Upo Committee Australia Ltd	liac liac	560.05	(3,702)	Ordinary
-May- ! 4	UBS Securities Australia Ltd	Sell	70,023	(7,5/0)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	28	(3)	Ordinary
1ay-14	UBS Securities Australia Ltd	Sell	897	(97)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	61,235	(6,620)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	3,793	(410)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	15,559	(1,682)	Ordinary
1ay-14	UBS Securities Australia Ltd	Sell	12,978	(1,403)	Ordinary
2-May-14	ILBS Securities Australia Itd	195	3 802	(411)	Ordinary
2-May-14	I IBS Securities Australia 1td)	1721	(186)	Ordinary
2-May-14	I IRC Convertion Ametralia 14d	Jes	11 100	(1) 2000	Ordinary
	Libe Countries Australia Libe	170	0012	(177)	Constant of
A Mari 14	Liber Committee Australia Lite	11-21	500	(CC)	Ordinary
2-Iviay-14	UBS Securities Australia Ltd	N	60,745	(/gc'q)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	765	(54)	Ordinary
1ay-14	UBS Securities Australia Lto	Sell	955,06	(5,442)	Ordinary
2 * 4m; 14	UBS Securities Australia Ltd	Sell	18,511	(2,012)	Ordinary
14.	UBS Securities Australia Ltd	Sel:	1,112	(581)	Orginary
2-May-14		Sell	4,625	(200)	Ordinary
ay-14	UBS Securities Australia Ltd	Sell	7,992	(864)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,240	(134)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	1,064	(115)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	2,453	(265)	Ordinary
-May-14	UBS Securities Australia Ltd	Sell	2,165	(234)	Ordinary
22-Mav-14	1985 Sporintias Australia Ltd		***************************************		
	TIT BIBLISTS SHITTING SUIT	2	9.250	(1 000)	Ordinary

22-May-14 UBS 22-May-14 UBS 22-May-14 UBS 23-May-14 UBS	JBS Securities Australia Ltd	<u> </u>			
\top	o seculities Australia Ltd	Colt	1/0//	(707)	Ordinary
t	UBS Securities Australia Ltd	ķ 3	15 377	(1 657)	Ordinary
	UBS Securities Australia I td	- E	17.673	(1,899)	Ordinary
Ť	UBS Securities Australia I to	3	14 570	(1,570)	Ordinan
T	is Securities Australia Ltd	las	9.540	(1 028)	Ordinary
Г	UBS Securities Australia Ltd	is.	35,236	(3,797)	Ordinary
Г	S Securities Australia Ltd	ES.	19,219	(1,071)	Ordinary
Г	UBS Securities Australia Ltd	ls.	3,703	(388)	Ordinary
22-May-14 UB	UBS Securities Australia Ltd	18.	2,905	(313)	Ordinary
2-May-14 UB	UBS Securities Australia Ltd	les.	2.255	(243)	Ordinary
Т	S Securities Australia Ltd	34	5.911	(637)	Ordinary
22-May-14 UB	UBS Securities Australia Ltd	las	5.800	(625)	Ordinary
Ť	UBS Securities Australia Ltd	3	1.479	(154)	Ordinary
Ť	IRS Securities Australia Ltd	100	0000	(1,060)	ording.
Ť	Securities Australia Ltd	100	076'6	(600,1)	Ordinary
1	S Securities Australia Ltd	i S	6-	(7)	Orginary
1	S Securities Abstraina Ltd	II S	931	(80)	Ordinary
	UBS Securities Australia Ltd	- Ne	2,441	(263)	Ordinary
1	DDS Securities Australia Ltd	is c	9/0/1	(911)	Ordinary
Ť	Jobs Securities Australia Ltd	3	2,830	(305)	Ordinary
Ť	JBS Securities Australia Ltd	- Kell	5,964	(750)	Ordinary
- 1	S Securities Australia Ltd	Sell	13,928	(1,500)	Ordinary
2-May-14 UB	Securities Australia Ltd	Sell	13,928	(1,500)	Ordinary
	UBS Securities Australia Ltd	llas.	9,285	(1,000)	Ordinary
- [S Securities Australia Ltd	Sell	7,428	(800)	Ordinary
1	JBS Securities Australia Ltd	Sell	122,024	(13,135)	Ordinary
T	UBS Securities Australia Ltd	38	3,762	(405)	Ordinary
- 1	S Securities Australia Ltd	Sell	942,695	(101,913)	Ordinary
- 1	UBS Securities Australia Ltd	lles	47,665	(5,153)	Ordinary
22-May-14 UB	S Securities Australia Ltd	Sell	2,553	(3/6)	Ordinary
- 1	S Securities Australia Ltd	Sell	5,874	(635)	Ordinary
Ť	JBS Securities Australia Ltd	Sell	1,082	(117)	Ordinary
1	JBS Securities Australia Ltd	ES.	8,695	(940)	Ordinary
- [JBS Securities Australia Ltd	Sell	10,998	(1,189)	Ordinary
ı	S Securities Australia Ltd	Ja .	1,768	(191)	Ordinary
22-May-14 UB	UBS Securities Australia Ltd	Sell	26,451	(2,858)	Ordinary
- {	S Securities Australia Ltd	Sell	3,830	(414)	Ordinary
Ī	UBS Securities Australia Ltd	Sell	4,986	(539)	Ordinary
_	UBS Securities Australia Ltd	Sell	15,170	(1,640)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	1,462	(158)	Ordinary
	S Securities Australia Ltd	Sell	1,637	(177)	Ordinary
2-May-14 UB	S Securities Australia Ltd	Sell	8,714	(642)	Ordinary
	UBS Securities Australia Ltd	Sell	2,951	(319)	Ordinary
l	UBS Securities Australia Ltd	les	196,1	(212)	Ordinary
	S Securities Australia Ltd	Sell	1,194	(129)	Ordinary
	S Securities Australia Ltd		16,243	(1,755)	Ordinary
	UBS Securities Australia Ltd	les	1,230	(133)	Ordinary
	S Securities Australia Ltd	Sell	11,711	(185)	Ordinary
	UBS Securities Australia Ltd	Sell	13,995	(1,513)	Ordinary
2-May-14 UB	UBS Securities Australia Ltd	Sell	11,849	(1,281)	Ordinary
	UBS Securities Australia Ltd	ll-8S	21,895	(2,367)	Ordinary
	S Securities Australia Ltd	Sell	1,711	(185)	Ordinary
l	UBS Securities Australia Ltd	Sell	694	(75)	Ordinary
	S Securities Australia Ltd	llas.	3,293	(356)	Ordinary
l	UBS Securities Australia Ltd	Sell	5,245	(267)	Ordinary
	UBS Securities Australia Ltd	les.	1,684	(182)	Ordinary
ı	S Securities Australia Ltd	llay.	19,647	(2,124)	Ordinary
1	S Securities Australia Ltd	lles	7,335	(793)	Ordinary
1	UBS Securities Australia Ltd	Sell	7,835	(847)	Ordinary
ı	UBS Securities Australia Ltd	lles	2,784	(301)	Ordinary
Ī	UBS Securities Australia Ltd	lles	14,881	(1,607)	Ordinary
Ī	JBS Securities Australia Ltd	Sell	989'9	(722)	Ordinary
_	UBS Securities Australia Ltd	Sell	2,797	(302)	Ordinary
Ē	S Securities Australia Ltd	Sell	1,296	(140)	Ordinary
✝¯	S Securities Australia 11d	3	73 474	(2535)	Ordinary
22-Ividy-14 UD	ODS Seculifies Australia Ltu	Jell Jell	4/4/57	(000,2)	Counary

Z-IMay-14	Lanco C	11-29	-		2000
May 14	Libs Cognition Australia Ltd	llay.	1 194	(120)	Ordinary
22-May-14	UBS Securities Australia Ltd	Jes Jes	386	(42)	Ordinary
22-May-14	I IRS Securities Australia I td	9	1156	(175)	Ordinary
1 - Val	LIBS Counties Australia Ltd	Zoll Zoll	0000	(67)	Gallary
41-VBIM-22	UBS Securities Australia Ltd	II-AC	230	(43)	Oraniary
Z-May-14	UBS Securities Australia Ltd	Sell	5/4	(7g)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seil	011,1	(071)	Ordinary
lay-14	UBS Securities Australia Ltd	Seil	934	(101)	Ordinary
22-May-14	UBS Securities Australia Ltd	Seli	1,073	(116)	Ordinary
22-May-14	UBS Securities Australia Ltd	Self	1,489	(191)	Ordinary
2-May-14	UBS Securities Australia Ltd	Seil	3,617	(391)	Ordinary
2 May 14	UBS Securities Australia Ltd	Seil	1.730	(187)	Ordinary
2-May-14	I IBS Cognitities Australia I to	Coll	963	(103)	Ordinary
	Day of the American Cook	100	100	(201)	or direct
22-IMay-14	UBS Securities Australia Ltd) all	101,1	(61)	Crainary
2-May-14	UBS Securities Australia Ltd	Sell	1,230	(133)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	2,118	(525)	Ordinary
2-May-14	UBS Securities Australia Ltd	Self	167	(18)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	19	(2)	Ordinary
4av-14	UBS Securities Australia Ltd	Sell	15,337	(1,658)	Ordinary
2-May-14	IUBS Securities Australia Ltd	35	1.545	(167)	Ordinary
22-May-14	LIBS Securities Australia Ltd	ll-9S	6.734	(778)	Ordinary
2-May-14	LIBS Securities Australia Ltd	Sell	1.489	(161)	Ordinary
May-14	I IBC Counties Australia I td	3	1 0.45	(113)	Ordinary
14	I IDC Countries Australia 1+d	3	1 023	(000)	Ordingo
2 Mar. 14	Let Glaster A collision of the	1100	1 115	(502)	Ordinary Ordinary
1dy-14	Obs securities Australia Ltd	100	014,-	(601)	Oranial y
2-IVIdy-14	UBS Securities Australia Ltd	IIAC II	040,01	(80/1)	Consulary Consulary
Z-May-14	UBS Securities Australia Ltd	Sell	897	(67)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	6,235	(674)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	953	(103)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	5,245	(295)	Ordinary
Jav-14	UBS Securities Australia Ltd	-SX	3.987	(431)	Ordinary
2-May-14	LIBS Securities Australia Ltd	Sell	1992	(83)	Ordinary
22-May-14	I IBS Securities Australia I to	19	15.448	(1 670)	Ordinary
2 May 14	In the case with the Australia Library	1103	10 10	(2/2/1)	Ordinary
1dy-14	Upo Securities Australia Ltd	llac C~II	505,51	(40)	Ordinary
Z-May-14	Obs Securities Australia Ltd	120	****	(0+)	Colliary
22-May- 4	UBS Securities Australia Ltd	Sell	755,1	(144)	Ordinary
1ay-14	UBS Securities Australia Ltd	Sell	1,22,1	(132)	Ordinary
22-May-14	UBS Securities Australia Ltd	<u></u>	45,451	(4,911)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	18,741	(2,025)	Ordinary
22-May-14	I IBS Securities Australia I to	llaS	6022	(651)	Ordinary
	I IDC Countition Australia 1+d	Sell	1,055	(114)	Ordinan
13y- 14	UBS Securities Australia Ltd	II N	ccn'i	(114)	Ordinary
Aay-14	UBS Securities Australia Ltd	Seil	130	(14)	Ordinary
22-May-14	UBS Securities Australia Ltd	Sell	64,602	(6,984)	Ordinary
22-Mav-14	UBS Securities Australia Ltd	Sell	92,500	(10,000)	Ordinary
2-May-14	I IBS Societies Australia I to	Cell	62 484	(6.755)	Ordinary
101	ODD Securities Australia Etc.	1-0	101,20	(5,00)	0
-May-14	UBS Securities Australia Ltd	Sea	63,492	(b,864)	Ordinary
Aay-14	UBS Securities Australia Ltd	Seli	15,207	(1,644)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	162,2	(929)	Ordinary
2-May-14	I IBS Securities Australia ! td	Sell	7.594	(821)	Ordinary
11	De Contrator O reitar per Coll	0	250.60	(OO) (J	Ordinan
4-1-API-1-7	Obs securities Australia Etc	361	50,62	(4,430)	Ordinary
Z-May-14	UBS Securities Australia Ltd	Sell	805,1	(103)	Ordinary
1ay-14	UBS Securities Australia Ltd	Sell	2,165	(234)	Ordinary
2-May-14	UBS Securities Australia Ltd	lleS	1,369	(148)	Ordinary
2-May-14	I IBS Securities Australia ! to	las	2 618	(283)	Ordinary
101	Late of control Acceptance and	ljeg.	20212	(1004)	Ordinary.
4-14PM-7	Opp Securities Australia Liu	IID?	710'64	(100'0)	Clumary
Z-May-14	UBS Securities Australia Ltd	Sell	15,364	(1,661)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	20,831	(2,252)	Ordinary
Aay-14	UBS Securities Australia Ltd	Sell	1,480	(160)	Ordinary
2-May-14	UBS Securities Australia Ltd	Sell	44.770	(4,840)	Ordinary
17	I IDC Convision Australia 14d	===	03.010	(720 8)	Ordinan
22-May-14	UBS Securities Australia Ltd	Sell	010,50	(0,974)	Cidirlary
2-May-14	UBS Securities Australia Ltd	Sell	5,781	(625)	Ordinary
2-Mav-14	UBS Securities Australia Ltd	Sell	15,364	(1,661)	Ordinary
2-May-14	I IBS Securities Australia I td	lleS	13.875	(1.500)	Ordinary
	1 DO Commission Assessed in the	*3	0300	(000 1)	Ordina
22-Mdy-14	UBS Securities Australia Ltd	nac.	20.2.0		
				(000'1)	Cidiriary
22-Mav-14	UBS Securities Australia Ltd	Sell	69'6	(1,042)	Ordinary

2 1.4 1.4	COO CIODAI Asset Ivialiageline it	- And	601,444,163	69,233	Ordinary
3-M3V-14	UBS Global Asset Management	218	44 971	4 876	Ordinary
3-May-14	Ť	Buy	82,197	8 834	Ordinary
23-May-14	ŧΞ	Buy	179.734	19.263	Ordinary
3-Mav-14	Ť	Transfer-In	Δ/N	1 820	Ordinary
23-Mav-14	ŧΞ	Transfer-In	Δ/N	1 829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AN	609	Ordinary
3-May-14	LIBS Wealth Management Australia Ltd	Transfectio	VIV	1 210	Ording
-May-14	Ť	Transfer-In	VIN	0081	Ordinary Ordinary
23-May-14	┪	Transfer-in	A/N	1 879	Ordinary
23-May-14	۳	Transfer-in	Ø/N	1 829	Ordina
23-May-14	UBS Wealth Management Australia Itd	Transfer-in	V V	1 210	Ordinal y
23-May-14	I IRS Wealth Management Australia Ltd	Transfer-In	VIV	003	Critical
8-May-14	LIRS Wealth Management Australia 14d	Transfer in		000	Graniary
A.May. 14	I IDC Worlds Management Australia Ltd	Transfer		670'1	Ordinary
11 - Kalay - C.	LIDS Wealth Mariagellefit Australia Ltd	T talisier-m	42	679'	Ordinary
Man 14	Ups wealth wanagement Australia Ltd	Iransier-In	¥.	678'1	Ordinary
3-May-14	Ubs wealth Management Australia Ltd	ranster-In	A/N	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝΆ	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,219	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Buy	3,114	336	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑM	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	₹ Z	1.829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AVA	1.879	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	A/N	1,829	Ordinary
3-May-14	LIBS Wealth Management Airstralia Ltd	Transfar-In	V/A	1 000	Ording
3-May 14	TIDE World # Alangament Australia 14d	Transfer II	V0.	670,1	Oldmary
, and	CDS Wealth Managernett Australia Ltd	II drisiei-in	2	500	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	I ranster-in	NA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	₹ <u>₹</u>	1.829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	₩.	1,879	Ordinary
3-May-14	I IRS Wealth Management Asstralia Ltd	Transfor-in	012	000	Ordinary
2-May-14	I BC Mealth Management Australia Ltd	Transfer la	0 00	000	Ordinary
J. May 17	COS Manufacture Manufacture Cost and Line	T C C	T.	670'	Guinaly
INION 14	ODS Wearth Management Australia Ltd	Halislet-III	¥2	6171	Ordinary
3-ividy-14	ODS WEGILLI MANAGEMENT AUSTRAIN LTD	Iransier-in	A.	678'1	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	121	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	₹N	1,829	Ordinary
3-Mav-14	UBS Wealth Management Australia Ltd	Transfer-In	AVA	1 219	Ordinary
3-May-14	LIBS Wealth Management Australia Ltd	Transfer-In	472	003	Caliba
, A	COS Wealth Manager Herrit Australia Ltd	il distriction	¥	500	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	¥≱	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	Α'N	609	Ordinary
3-May-14	LIRS Wealth Management Australia Ltd	Transfer-In	NV	1 820	, decipato
A 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	LIDE World Management Australia Ind	T	C ***	670'	0
May 14	ODS Wealth Maliagement Australia Ltd	II-IIII	4	SOO ,	Ordinary
-ividy- i 4	Obs wealth imanagement Australia Ltd	l ransrer-in	₹.2	1,219	Ordinary
-May-14	UBS Wearth Management Australia Ltd	I ranster-in	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-in	N/A	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΥN	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-in	¥Ž	1.219	Ordinary
3-Mav-14	UBS Wealth Management Australia Ltd	Transfer-in	Α/N	1 879	Ordinary
May 17	i IRC Mosth Management Australia Ltd	Transfer In	V 1	020,1	ording.
M-14	Little Month Management Australia	Tenester in		000	A IIII
3-Ividy-14	Upp wealth management Australia Ltd	Iranster-tn	Ž.	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-in	N/A	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-in	ΝA	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΥN	609	Ordinary
3-Mav-14	UBS Wealth Management Australia Ltd	Transfer-In	Α/N	1 829	Ordinary
May 14	I IBC Wealth Management Australia Ind	Transfer In	V/14	000	ording.
- APIA-1-	Upp Wealth Mariagement Australia Ltd	וושואובווו	YA!	500	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transter-In	NA	1,829	Ordinary
-May-14	IUBS Wealth Management Australia Ltd	Transfer-In	VIV	101	7-0
				171	Ordinary
3-Mav-14	UBS Wealth Management Australia Ltd	Transfer-In	Ψ.N	609	Ordinary

1			067		
1	UBS Securities Australia I to	Sell	785 5	(577)	Ordinary
	UBS Securities Australia Ltd	Sell	953	(103)	Ordinary
22-May-14 UBS S	ecurities Australia Ltd	35	22.228	(7.403)	Ordinary
1	Securities Australia Ltd	الع	2 433	(263)	Ordinary
Т	I BS Securities Australia 11d	3	2 997	(324)	Ordinary
Т	Securities Australia 11d	3	2 932	(217)	Ordinary
Ť	on rities Australia Itd	و	2 030	(7:5)	Ordinary
220 41 100 22	DO Octobrica Action Ltd		2,233	(010)	Oldina.
۲	Continues Australia Ltd	100	616,6	(000)	Guinary
Ť	Jos Securities Australia Liu) dell	2,788	(040)	Ordinary
Ť	UBS Securities Australia Ltd	ja v	068'/	(852)	Ordinary
7	JBS Securities Australia Ltd	Sell	1,065	(115)	Ordinary
	UBS Securities Australia Ltd	Sell	4,084	(441)	Ordinary
2-May-14 UBS 54	ecurities Australia Ltd	Sell	17,853	(1,928)	Ordinary
_	UBS Securities Australia Ltd	les.	18,159	(1.961)	Ordinary
	Politips Australia 11d	ll e2	306	(55)	vicaion
T	IDC Counting Australia 14d	100	2300	(010)	Ordinary Ordinary
T	ברת וונפס עמפון שום דומ	ii ac	066,2	(607)	Oldinary
7	Securities Australia Ltd	Seil	2,401	(657)	Ordinary
22-May-14 UBS So	ecurities Australia Ltd	Sell	3,315	(328)	Ordinary
_	UBS Securities Australia Ltd	Sell	6,417	(693)	Ordinary
	UBS Securities Australia Ltd	Seli	08'8	(932)	Ordinary
1	ecurities Australia I td	las	1 101	(110)	Ordinan
2000 11 100 000	DO Commission Access (4-4)	100	000 10	(C) 1)	Graniary
+	ecurines Australia Ltd	100	909,62	(7,790)	Ordinary
7	Securities Australia Ltd	Sell	8,048	(870)	Ordinary
٦	ecurities Australia Ltd	Sell	9,722	(1,051)	Ordinary
	UBS Securities Australia Ltd	Sell	2,701	(262)	Ordinary
Г	UBS Securities Australia Ltd	Sell	2,673	(583)	Ordinary
Γ	ecurities Australia Ltd	-8×	1.008	(109)	Ordinary
T	UBS Securities Australia Itd	195	24 624	(2,662)	Ordinary
T	IDC Counting Australia 14d	0	4 410	(477)	Ordinon
Ť	erunines Australia Liu		214/4	(1)	Oldinary
┪	ecurities Australia Ltd	Sell	1,073	(116)	Ordinary
22-May-14 UBS Se	UBS Securities Australia Ltd	lle Sell	490	(23)	Ordinary
Ť	UBS Securities Australia Ltd	Sell	9,250	(1,000)	Ordinary
Ť	IBS Securities Australia Ltd	la ₂	4 246	(450)	Ordinan
Ť	DO COMPANY AND THE PARTY OF THE	11-0	047,4	(604)	Collinary
1	ברחווובי אחצוו פוים דות	261	200,0	(304)	Ordinary
┪	UBS Securities Australia Ltd	Sell	16,798	(1,816)	Ordinary
22-May-14 UBS Se	UBS Securities Australia Ltd	lles)	13,709	(1,482)	Ordinary
Ē	UBS Securities Australia Ltd	Sell	3,728	(403)	Ordinary
T	JBS Securities Australia Ltd	195	6 207	(671)	Ordinac
Ť	IBS Secripting Australia 11d	le)	15 566	(300 1)	Ording
Ť	IRC Countries Australia 11d	(Sell	2 442	(450)	Ordinary
Ť	ecolities Aostralia Liu	100	2447	(407)	Galliary
-May-14 UBS S6	UBS Securities Australia Ltd	Sell	87,533	(9,463)	Ordinary
┪	ecurities Australia Ltd	Sell	9,278	(1,003)	Ordinary
_	UBS Securities Australia Ltd	Sett	17,964	(1,942)	Ordinary
22-May-14 UBS Se	UBS Securities Australia Ltd	Sell	9,222	(266)	Ordinary
Ī	UBS Securities Australia Ltd	Sell	26.104	(2,822)	Ordinary
T	IBS Securities Australia I to	Ş	3 101	(345)	Ordinary
T	Scalines Australia Ltd	100	507.55	(240)	O-diany
22-Way-14	Obs seculities Australia Liu	100	507,703	(9/0/4)	Clarray
1	ecurities Australia Ltd	Sell	078'9	(/38)	Ordinary
23-May-14 UBS Se	UBS Securities Australia Ltd	Buy	124,874	13,594	Ordinary
23-May-14 UBS Se	UBS Securities Australia Ltd	Sell	65,267	(10,530)	Ordinary
Т	UBS Sequitities Australia Ltd	Bis	137 698	14 394	Ordinary
T	bt cileaton A ocities po	00	07.761	(0000)	Carding
20-(viay-14 000 00	ODD DECULINES AUSTRALIA	iibc	10/'/8	(2,402)	Colliary
7	ecurities Australia Ltd	Buy	185,250	411,02	Ordinary
23-May-14 UBS Se	UBS Securities Australia Ltd	Sell	9,270	(1,000)	Ordinary
	UBS Global Asset Management	Buy	327,996	35,459	Ordinary
Γ	JBS Global Asset Management	Riv	3 693	397	Ordinary
T	Global Assat Management	Bin	E 129 257	551 152	Ordinan
3	ODS GIODAL ASSET INTALIABELIEUR	, in	162,631,0	20,100	o di ilia
1	lobal Asset Management	Buy	1,/39,8/1	186,991	Ordinary
23-May-14 UBS G	UBS Global Asset Management	Buy	159,884	17,180	Ordinary
23-May-14 UBS G	JBS Global Asset Management	Buy	165,868	17,823	Ordinary
T	IRS Global Asset Management	25.5	364 440	39 160	Ordinary
T	lakal Asset Managaritan	62.6	100,000	20,000	O'clinaly O'clinaly
3-May-14 UBS G	UBS Giobal Asset Management	Buy	1,582,605	170,055	Ordinary
	fobal Asset Management	Buy	1,367,626	146,955	Ordinary
Г	John Asset Management	Bin	3 515 764	877 775	Ordinary
Т	UBS Global Asset Ivianagement	Buy	1,010,004	3//'//8	Ordinary
Г	Labor Accast Management	× is	130	37 A CC	vacainan

47-ividy 14	מכי זויבוויו אופוופלבוויבווי שמנופום דום				
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	I ransier-in	N/A	1,829	Ordinary
23-IVIdy-14	UBS wealth Management Australia Ltd	I ransfer-in	A'A	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝΑ	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	A/N	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AVA	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AVA	1.829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
23-Mav-14	Wealth Management	Transfer-In	A/N	509	Ordinan
23-May-14	UBS Wealth Management Australia 11d	Transfer.in	VIV	000 \$	in a constant
23-May-14	I IRS Wealth Management Australia itd	Transfer.In		0001	y is is
23-May-14	I IRC Wealth Management Australia the	Transfer in	¥ .	670'1	Ordinary
23-VIOY-14	Obs weath management Australia Ltd	I ranster-in	AN.	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	ranster-tn	AA A	1,829	Ordinary
23-IVidy-14	UBS Wealth Management Australia Ltd	Iranster⊣n	AN	1,219	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA AA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AN	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝΑ	121	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AVA	1.219	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	¥X	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1.829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AVA	1 879	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1 879	Ordinary
23-May-14	UBS Wealth Management Australia 11d	Transfer-In	AVA	1 870	Ordinary
23-May-14	I IBC Wasith Management A. straigs 144	Transfer	4/14	620,1	Ordinary
1000	1 DC MAnaget Management A service 14-1	1112121111	Y/A:	600	Oldinary
23 May 14	UBS Wealth Management Australia Ltd	r c	WA	617,1	Ordinary
23-May-14		Iranster-in	N/A	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	I ranster-In	AA A	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-tn	N/A	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-in	A/A	121	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-in	AW	121	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-in	AVA	1.829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-in	AVA	121	Ordinary
23-May-14	I IRC Woolth Management Australia 14d	Transfor in	C 47.4	120	Ordanary Ordanary
23-May 14	Mealth Management	Transfer In		570,1	S S S S S S S S S S S S S S S S S S S
22 May 14	1 IDS Washing Management Australia to	Topological	X 2	171	Galnary
23 May 14	Obs wealth Management Australia Ltd	ı ransrer-in	NA.	171	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	l ranster-In	NA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	A/A	1,219	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-in	ΝΑ	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	Ϋ́	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	Ž	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	Ϋ́	1.829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	1,219	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	Ϋ́N	1.829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	Ϋ́	1,219	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	1.829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	₹Ž	121	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	1.829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	Ϋ́	609	Ordinary
23-Mav-14	UBS Wealth Management Australia Ltd	Transfer-In	A/A	609	Ordinary
23-Mav-14	(IRS Wealth Management Australia 11d	Transfer-In	VIV	1 829	Ordinary
23-May-14	UBS Wealth Management Australia (1d	Transfer-in	AVA	009	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AVA	1 870	Ordinary
23-May-14	I IRS Wealth Management Australia 11d	Transfer-in	C VIV	1 020	Ordinary
23-May-14	I IBS Wealth Management Australia Ind	Transfords	X X	670'1	Ordinary
23-Ividy-14	UBS Wealth Management Australia Lid	ransier-in	AN.	678'1	Ordinary
23 May 14	UBS Wealth Management Australia Ltd	Transfer-In	NA.	509	Ordinary
23-Mdy-14	Obs wealth Management Australia Ltd	Iransfer-In	WA	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	V// V		
			A.M.	1,829	Ordinary

23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1.219	Crdinary
23-Mav-14	UBS Wealth Management Australia Ltd	Transfer-In	4/2	1 879	Ordinary
23-May-14	I ISC Woolth Management Australia Ltd	Transfer in	474	003	Ordinary
	DO WE ALL MANAGEMENT AUSTRALIA LICE	1		500	Ordinary
+1-(a)-(-2	ODS Wealth Management Australia Ltd	I dan Sier-Ri	2	670'	Ordinary
23-IVIdy-14	Upo Wealth Management Australia Ltd	I ransier-in	Z/A	600	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	l ranster-In	ΔN.	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Iranster-In	ν. V.	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	121	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	₹X	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AVV	609	Ordinary
3-May-14	I BS Wealth Management Australia Ltd	Transfer-In	S/N	1 870	Ording
13-May-14	I BC Mealth Management Australia I to	Transfor In	4712	1 240	Cardinary
1	COS Wealth Management Australia Etc	1	17.7	617/1	Cicinary
3-IVIdy-14	Obs wealth Management Australia Ltd	i ranster-in	¥N	600	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-tn	NA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
23-Mav-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	1 829	Ordinary
3-May-14	I IBS Wealth Management Australia 14d	Transfer-In	SIZ	1 020	Croling
- May	Obstantia Management Australia Ltd	I disiel-III	XA.	1,029	Croinary
23-INIdy-14	Obs wealth Management Australia Ltd	Iransrer-In	NA.	600	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transter-In	¥ Ž	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA.	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	1.219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-in	AVA	509	Ordinary
A May 14	I IBC Month Management Assets in the	Transforts	4	000	i i i
	and the second s	III DISTIBLE		3	Clumary
3-IVIdy- 14	Upo Wealth Management Australia Ltd	nansiel∹in	NA.	617,1	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΥA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AN	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	¥2	1.829	Ordinary
23-May-14	LiRS Wealth Management Australia Ltd	Transfer-In	W.W.	1 820	Ordingo
2 140	100 Mar - Ma	-1 -9		570,1	, G. G. G.
3-May-14	UBS Wearth Management Australia Ltd	I ranster-in	A'N	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΥN	1.829	Ordinary
3-May-14	I IBS Wealth Management Australia Ltd	Transfar-In	W.	1 820	Ordinan
5 May 14	I IDC Washing Management Assessment Land			020.	Cicilian y
2-Iviay-1-	Opp wealth management Australia Ltd	I di islet-ii i	۲ <u>۶</u>	670'	Ordinary
S-Ividy 14	UBS Wealth Management Australia Ltd	I ransier-in	A/A	617,1	Ordinary
3-May-14	IUBS Wealth Management Australia Ltd	iranster-In	NA	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	₹X	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AN	1.879	Ordinary
3-May-14	It BC Manith Management Australia Ltd	Transfer-In	A14A	1 020) adiabatic
5 May 14	I DC Western Internagement Australia Ltd	Transfer in	YA:	270'	Clumary
3-jvlay-14	ODS Wealth Management Australia Ltd	I drister-in	2	678'	Ordinary
3-May-14	IUBS Wealth Management Australia Ltd	Transfer-In	×Ν	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	Ϋ́	1.829	Ordinary
3-May-14	LIRS Wealth Management Australia Ltd	Transfer-In	Ø/N	609	Ordinan
2 14-2	I DC WALLER MALE TO THE AMERICAN COLOR	1		3	Oldinary Clarinary
3-Wdy-14	Obs wealth Management Australia Ltd	ıransıer-ın	Y.	678'	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	i ranster-In	ΑN	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia i to	Transfer-In	∇/N	1 879	Ordinary
A.May. 14	I IBC Weelth Management Australia ! to	Tranefer-In	V/4	1 210	Ording
- Kildy	ביים אגבשונו ואושוים לביווים וויים ביים	T Calculation	1	6171	Gollaly
3-May-14	UBS Wealth Management Australia Ltd	Iranster-In	ΔN	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΨN	1.219	Ordinary
3.May.14	1 (85 Wealth Management Australia 1 td	Transfer-In	VIV	000	Ordina
1	COS Predict Management Australia COS	1 0 0 0	(9)	670'	Ciciliary
3-May-14	UBS Wealth Management Australia Ltd	I ranster-in	NA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,219	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	¥×	609	Ordinary
3-May-14	LIRS Wealth Management Australia Ltd	Transfer-In	Φ/N	509	Ordinary
3-May-14	I IBC Manith Management Australia 1 to	Transfer-In	VIV	1 000	Ordinan
- Ividy 14	COS Wealth Management Australia Ltd	110113161-111	Z	670,	Clustery
3-May-14	UBS Wealth Management Australia Ltd	ı ranster-in	NA VA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-in	ΑΝ	1.829	Ordinary
3-May-14	LIBS Wealth Management Australia Ltd	Transfer-In	AVA	1 879	Ordinary
22 8400, 14				1 have	
3-May-14		, ,			(10,11,11,1
, (p)	UBS Wealth Management Australia Ltd	Transfer-In	AN	1,829	Ordinary

BS Western Management Australia LIG Transfer-In NAA 1,829 BS Western Management Australia LIG Transfer-In NAA 1,829 BS Western Management Australia LIG Transfer-In NAA 1,219	20 144. 14	The Control of the second of the Control of the Con				
Bits Wealth Management Australia Ltd Transfer-in NAM 1,229 168 Wealth Management Australia Ltd Transfer-in NAM 1,219 168 Wealth Management Australia Ltd Transfer-in NAM 1,229	-May-14	UBS Wealth Management Australia Ltd	Transfer-In	(A/N	1 879	Ordinary
UBS Wealth Nanogement Australia Ltd Transfer-in NAM 1,829 UBS Wealth Nanogement Australia Ltd Transfer-in NAM 1,219 UBS Wealth Nanogement Australia Ltd Sell 2,210 (1,27) UBS Securities Australia Ltd Sell 2,210 (2,20) UBS Secu	-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑΝ	1.829	Ordinary
185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Transfer-in NA 1,219 185 Weeth Management Australie Ltd Sell Transfer-in NA 1,219 185 Weeth Management Australie Ltd Sell Transfer-in NA 1,219 185 Securies Australie Ltd Sell Transfer-in NA 1,219 185 Securies Australie Ltd Sell Transfer-in NA 1,219 185 Securies Australie Ltd Sell Transfer-in NA 1,219 185 Securies Australie Ltd Sell Sell 1,229 1,220 185 Securies Australie Ltd Sell Sell 2,286 1,220 1,220 185 Securies Australie Ltd Sell Sell 2,286 1,220 1,220 185 Securies Australie Ltd Sell Sell 2,286 1,220 1,220 185 Securies Australie Ltd Sell Sell 2,286 1,220 1,220 185 Securies Australie Ltd Sell Sell 2,286 1,220 1,220 185 Securies Australie Ltd Sell Sell 3,260 1,200 1,200 185 Securies Australie Ltd Sell 3,260 1,200 1,200 185 Securies Australie Ltd	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	4/N	1 879	Ordinary
155 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 156 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 157 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 158 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 158 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 158 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 158 Wealth Management Australia Ltd Transfer-in NA 1, 12/19 158 Securities Australia Ltd Sell Transfer-in NA 1, 12/19 158 Securities Australia Ltd Sell Transfer-in NA 1, 12/19 158 Securities Australia Ltd Sell Sell 2,250 (250) 158 Securities Australia Ltd Sell Sell 2,250 (250) 158 Securities Australia Ltd Sell Sell 2,260 (250) 158 Securities Australia Ltd Sell Sell 2,260 (250) 158 Securities Australia Ltd Sell 2,260 (250) (250) 158 Securities Australia Ltd Sell 2,260 (250) (250) 158 Securities Australia Ltd Sell 2,260 (250) (250) 158 Securities Australia Ltd Sell 2,260 (250) (250) 158 Securities Australia Ltd Sell 2,260 (250) (250) 158 Securities Australia	3-May-14	I IRS Wealth Management Australia 11d	Transfer-In	VIV	0101	ording.
USS Wealth Management Australia Ltd Transfer-in NAM 1,239 USS Wealth Management Australia Ltd Transfer-in 15,294 1,239 USS Wealth Management Australia Ltd Transfer-in NAM 1,239 USS Wealth Management Australia Ltd Transfer-in NAM 1,829 USS Wealth Management Australia Ltd Transfer-in NAM 1,829 USS Wealth Management Australia Ltd Transfer-in NAM 1,829 USS Wealth Management Australia Ltd Transfer-in NAM 1,219 USS Wealth Management Australia Ltd Sell 2,345 1,270 USS Wealth Management Australia Ltd Sell 2,345 1,270	3-May-14	URS Wealth Management Australia Ltd	Transfer-In	Ç. VIN	009	Ordinary Ordinary
LBS Wealth Management Australia Ltd Transfer-in NWA 1,219 LBS Wealth Management Australia Ltd I Transfer-in NAA 669 LBS Wealth Management Australia Ltd Transfer-in NAA 669 LBS Wealth Management Australia Ltd Transfer-in NAA 1689 LBS Wealth Management Australia Ltd Transfer-in NAA 1689 LBS Wealth Management Australia Ltd Transfer-in NAA 1689 LBS Wealth Management Australia Ltd Transfer-in NAA 17219 LBS Wealth Management Australia Ltd Transfer-in NAA 17219 LBS Wealth Management Australia Ltd Transfer-in NAA 17219 LBS Wealth Management Australia Ltd Transfer-in NAA 1,219 LBS Wealth Management Australia Ltd Sell 2,415 1,529 LBS Securities Australia Ltd Sell 2,415 1,629 LBS Securities Australia Ltd Sell 2,315 1,629 LBS Securities Australia Ltd Sell 2,316 1,629 LBS Securities Australia Ltd	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	A/N	1 219	Ordinary
UBS Vealth Namegement Australia Ltd Transfer-in NAM 1,519 UBS Vealth Namegement Australia Ltd Transfer-in NAM 1,529 UBS Vealth Namegement Australia Ltd Transfer-in NAM 1,829 UBS Vealth Namegement Australia Ltd Transfer-in NAM 1,829 UBS Vealth Namegement Australia Ltd Transfer-in NAM 1,521 UBS Securities Australia Ltd Sell 2,325,000 (8,290) UBS Securities Australia Ltd Sell 2,325,000 (8,290) UBS Securities Australia Ltd Sell 2,325,000 (6,70) UBS Securities Australia Ltd Sell 2,326 (7,00) UBS Securities Australia Ltd Sell 2,326 (7,00) UBS Securities Australia Ltd	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AN	1,219	Ordinary
(DES Weathir Narianalis LID Buy 15,624 1,686 (DES Weathir Narianalis LID Transfer-In NA 1,687 (DES Weathir Narianalis LID Transfer-In NA 1,687 (DES Weathir Narianalism Augustine LID Transfer-In NA 1,829 (DES Weathir Narianalism Augustine Lid Transfer-In NA 1,829 (DES Weathir Narianalism Augustine Lid Transfer-In NA 1,219 (DES Weathir Narianalism Augustine Lid Transfer-In NA 1,219 (DES Weathir Narianalism Lid Transfer-In NA 1,219 (DES Weathir Narianalism Lid Transfer-In NA 1,219 (DES Weathir Narianalism Lid Transfer-In NA 1,219 (DES Weathir Augustine Lid Transfer-In NA 1,219 (DES Weathir Augustine Lid Transfer-In NA 1,219 (DES Securities Augustine Lid Transfer-In NA 1,219 (DES Securities Augustine Lid Sell 2,320 (7,20) (DES Securities Augustine Lid Sell 2,320 <td>3-May-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Transfer-In</td> <td>AW</td> <td>1 219</td> <td>Ordinary</td>	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	AW	1 219	Ordinary
BES Weath Narabala Ltd Transfer-In NA 1,829 LBS Weath Narabarment Australia Ltd Transfer-In NA 1,829 LBS Weath Narabarment Australia Ltd Transfer-In NA 1,829 LBS Weath Narabarment Australia Ltd Transfer-In NA 1,219 LBS Securities Australia Ltd Sell 2,316 1,529 LBS Securities Australia Ltd Sell 2,320 1,618 LBS Securities Australia Ltd Sell 2,320 1,618 LBS Securities Australia Ltd Sell 2,320 1,620 LBS Securities Australia Ltd Sell 2,320 1,620 LBS Securities Australia Ltd Sell 2,320 </td <td>3-May-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Buv</td> <td>15.624</td> <td>1.686</td> <td>Ordinary</td>	3-May-14	UBS Wealth Management Australia Ltd	Buv	15.624	1.686	Ordinary
BSS Veckith Naragement Australia Ltd Transfer-in NA 1,829 BSS Veckith Naragement Australia Ltd Transfer-in NA 1,829 BSS Veckith Naragement Australia Ltd Transfer-in NA 1,219 BSS Veckithes Australia Ltd Sell Transfer-in NA 1,219 BSS Securities Australia Ltd Sell Transfer-in NA 1,219 BSS Securities Australia Ltd Sell	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	609	Ordinary
UBS Swetch Navagement Australia Ltd Transferin NA 1,829 UBS Weath Navagement Australia Ltd Transferin NA 1,219 UBS Weath Navagement Australia Ltd Transferin NA 1,219 UBS Swetch Management Australia Ltd Transferin NA 1,219 UBS Swetch Navagement Australia Ltd Transferin NA 1,219 UBS Swetch Navagement Australia Ltd Transferin NA 1,219 UBS Swetchires Australia Ltd Transferin NA 1,221 UBS Swetchires Australia Ltd Sell 23,265 16,800 UBS Securities Australia Ltd Sell 23,265 15,800 UBS Securities Australia Ltd Sell 23,265 16,800 UBS Securities Australia Ltd Sell 23,250 16,800 UBS Securities Australia Ltd Sell 2,255 17,00 UBS Securities Australia Ltd Sell 2,356 17,00 UBS Securities Australia Ltd Sell 2,356 17,00 UBS Securities Australia Ltd Sell 2,366 1	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	×N	1.829	Ordinary
UBS Wealth Management Australia Ltd Transfer-In NA C00 UBS Wealth Management Australia Ltd Transfer-In NA 1,219 UBS Wealth Management Australia Ltd Transfer-In NA 1,219 UBS Wealth Management Australia Ltd Transfer-In NA 1,219 UBS Wealth Management Australia Ltd Sell 1,229 1,219 UBS Wealth Management Australia Ltd Sell 1,239 1,219 UBS Securities Australia Ltd Sell 1,239 1,229 UBS Securities Australia Ltd Sell 2,335,00 1,204 UBS Securities Australia Ltd Sell 1,204 1,004 UBS Securities Australia Ltd Sell 1,204 1,004 UBS Securities Australia Ltd Sell 1,204 1,004 UBS Securities Australia Ltd Sell 1,104 1,004 UBS Securities Australia Ltd Sell 2,310 (1,220 UBS Securities Australia Ltd Sell 2,325 (1,005 UBS Securities Australia Ltd Sell 2,114 (1,220 <td>3-May-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Transfer-In</td> <td>ΑN</td> <td>1,829</td> <td>Ordinary</td>	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑN	1,829	Ordinary
UBS Swellth Management Australia Ld Transfer-In NA 6.05 UBS Wealth Management Australia Ld Transfer-In NA 1,21 UBS Wealth Management Australia Ld Transfer-In NA 1,22 UBS Wealth Management Australia Ld Transfer-In 1,52 1,82 UBS Wealth Management Australia Ld Sell 23,50 (1,820) UBS Securities Australia Ld Sell 23,50 (2,500) UBS Securities Australia Ld Sell 23,50 (2,500) UBS Securities Australia Ld Sell 1,51 (1,00) UBS Securities Australia Ld Sell 1,50 (1,00) UBS Securities Australia Ld Sell 1,50 (1,00) UBS Securities Australia Ld Sell 1,11 (1,00) UBS Securities Australia Ld Sell 2,30 (2,00) UBS Securities Australia Ld Sell 2,30 (2,00) UBS Securities Australia Ld Sell 2,30 (2,00) UBS Securities Australia Ld Sell 2,30 (2,00) <trr< td=""><td>3-May-14</td><td>UBS Wealth Management Australia Ltd</td><td>Transfer-In</td><td>ΝN</td><td>609</td><td>Ordinary</td></trr<>	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝN	609	Ordinary
UBS Sweath Management Australa Led Transfer-In NVA 1,219 UBS Wealth Management Australa Led Transfer-In NVA 1,219 UBS Wealth Management Australa Led Transfer-In NVA 1,219 UBS Sworthes Australia Ltd Sell 1,23,016 (E.829) UBS Securities Australia Ltd Sell 24,152 (C.590) UBS Securities Australia Ltd Sell 24,152 (C.590) UBS Securities Australia Ltd Sell 2,335 (C.590) UBS Securities Australia Ltd Sell 2,454 (T.57) UBS Securities Australia Ltd Sell 1,454 (T.57) UBS Securities Australia Ltd Sell 2,911 (T.67) UBS Securities Australia Ltd Sell 2,911 (T.67) UBS Securities Australia Ltd Sell 2,911 (T.67) UBS Securities Australia Ltd Sell 2,914 (T.67) UBS Securities Australia Ltd Sell 2,724 (T.67) UBS Securities Australia Ltd Sell 2,726 (T.07) <td>3-May-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Transfer-In</td> <td>N/A</td> <td>609</td> <td>Ordinary</td>	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	609	Ordinary
UBS Securities Australia Ltd Transfer-In IVA 1,219 UBS Wealth Management Australia Ltd Sell 1,521 (18,29) UBS Securities Australia Ltd Sell 23,512 (18,260) UBS Securities Australia Ltd Sell 23,502 (2,506) UBS Securities Australia Ltd Sell 23,500 (5,207) UBS Securities Australia Ltd Sell 1,617 (1,417) UBS Securities Australia Ltd Sell 1,627 (1,677) UBS Securities Australia Ltd Sell 1,627 (1,737) UBS Securities Australia Ltd Sell 1,677 (1,737) UBS Securities Australia Ltd Sell 1,677 (1,737) UBS Securities Australia Ltd Sell 2,935 (1,677) UBS Securities Australia Ltd Sell 2,943 (1,677) UBS Securities Australia Ltd Sell 2,943 (3,700) UBS Securities Australia Ltd Sell 2,748 (800) UBS Securities Australia Ltd Sell 2,748 (3,700) <tr< td=""><td>3-May-14</td><td>UBS Wealth Management Australia Ltd</td><td>Transfer-In</td><td>N/A</td><td>1,219</td><td>Ordinary</td></tr<>	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,219	Ordinary
UBS Securities Australia Ltd Transfer-In NA 1,229 UBS Swellth Management Australia Ltd Sell 1,52,12 (1,8,240) UBS Securities Australia Ltd Sell 2,3,30 (5,597) UBS Securities Australia Ltd Sell 2,3,30 (5,597) UBS Securities Australia Ltd Sell 2,3,50 (5,597) UBS Securities Australia Ltd Sell 1,454 (5,597) UBS Securities Australia Ltd Sell 1,454 (7,577) UBS Securities Australia Ltd Sell 1,454 (1,677) UBS Securities Australia Ltd Sell 2,911 (1,677) UBS Securities Australia Ltd Sell 2,911 (1,427) UBS Securities Australia Ltd Sell 2,736 (7,007)	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	121	Ordinary
UBS Securities Australia Ltd Tennifer-In T/5 2/15 (B-500) UBS Securities Australia Ltd Sell 22,306 (2,509) UBS Securities Australia Ltd Sell 22,152 (2,509) UBS Securities Australia Ltd Sell 22,150 (6,800) UBS Securities Australia Ltd Sell 1,518 (1734) UBS Securities Australia Ltd Sell 1,518 (1734) UBS Securities Australia Ltd Sell 1,518 (1737) UBS Securities Australia Ltd Sell 1,518 (1737) UBS Securities Australia Ltd Sell 2,033 (852) UBS Securities Australia Ltd Sell 2,033 (873) UBS Securities Australia Ltd Sell 2,033 (873) UBS Securities Australia Ltd Sell 2,034 (1,057) U	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,219	Ordinary
UBS Securities Australia Ltd Sell 772,12 (18,80) UBS Securities Australia Ltd Sell 23,152 (2,597) UBS Securities Australia Ltd Sell 23,152 (2,597) UBS Securities Australia Ltd Sell 1,618 (174) UBS Securities Australia Ltd Sell 1,618 (174) UBS Securities Australia Ltd Sell 1,752 (5,507) UBS Securities Australia Ltd Sell 1,744 (108) UBS Securities Australia Ltd Sell 1,711 (120) UBS Securities Australia Ltd Sell 1,711 (120) UBS Securities Australia Ltd Sell 2,714 (120) UBS Securities Australia Ltd Sell 2,736 (700) UBS Securities Australia Ltd Sell 2,744 (120) UBS Securities Australia Ltd Sell 5,256 (700) UBS Securities Australia Ltd Sell 5,256 (700) UBS Securities Australia Ltd Sell 5,256 (700) UBS Securities Au	3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
UBS Securities Australia Ltd Sell 2,3,506 (2,590) UBS Securities Australia Ltd Sell 24,356 (5,800) UBS Securities Australia Ltd Sell 1,004 (108) UBS Securities Australia Ltd Sell 1,004 (108) UBS Securities Australia Ltd Sell 1,004 (108) UBS Securities Australia Ltd Sell 1,111 (1,20) UBS Securities Australia Ltd Sell 1,311 (1,120) UBS Securities Australia Ltd Sell 9,093 (882) UBS Securities Australia Ltd Sell 9,131 (142) UBS Securities Australia Ltd Sell 2,174 (123) UBS Securities Australia Ltd Sell 2,174 (123) UBS Securities Australia Ltd Sell 2,174 (254) UBS Securities Australia Ltd Sell 2,174 (240) UBS Securities Australia Ltd Sell 2,174 (240) UBS Securities Australia Ltd Sell 2,174 (240) UBS Securities Aus	3-May-14	UBS Securities Australia Ltd	Sell	175,212	(18,840)	Ordinary
UBS Securities Australia Ltd Sell 24,152 (3.597) UBS Securities Australia Ltd Sell 772 (580) UBS Securities Australia Ltd Sell 1,618 (174) UBS Securities Australia Ltd Sell 1,618 (174) UBS Securities Australia Ltd Sell 1,711 (120) UBS Securities Australia Ltd Sell 2,911 (142) UBS Securities Australia Ltd Sell 2,911 (142) UBS Securities Australia Ltd Sell 2,714 (123) UBS Securities Australia Ltd Sell 2,748 (300) UBS Securities Australia Ltd Sell 2,748 (300) UBS Securities Australia Lt	3-May-14	UBS Securities Australia Ltd	Sell	23,306	(2,506)	Ordinary
UBS Securities Australia Ltd Sell 73.940 (5.80) UBS Securities Australia Ltd Sell 77.7 (880) UBS Securities Australia Ltd Sell 1,618 (174) UBS Securities Australia Ltd Sell 1,651 (172) UBS Securities Australia Ltd Sell 2,325,000 (500) UBS Securities Australia Ltd Sell 1,111 (172) UBS Securities Australia Ltd Sell 2,903 (682) UBS Securities Australia Ltd Sell 2,911 (172) UBS Securities Australia Ltd Sell 2,903 (682) UBS Securities Australia Ltd Sell 2,903 (682) UBS Securities Australia Ltd Sell 2,374 (254) UBS Securities Australia Ltd Sell 2,389 (700) UBS Securities Australi	3-May-14	UBS Securities Australia Ltd	Sell	24,152	(2,597)	Ordinary
UBS Securities Australia Ltd Sell 772 (83) UBS Securities Australia Ltd Buyl 1,624 (104) UBS Securities Australia Ltd Sell 2,235,000 (250,000) UBS Securities Australia Ltd Sell 1,454 (107) UBS Securities Australia Ltd Sell (57) (67) UBS Securities Australia Ltd Sell (67) (700) UBS Securities Australia Ltd Sell (570) (700) UBS Securities Australia Ltd	3-May-14	UBS Securities Australia Ltd	Seli	53,940	(2,800)	Ordinary
UBS Securities Australia Ltd Sell 1,618 (174) UBS Securities Australia Ltd Sell 1,544 (175) UBS Securities Australia Ltd Sell 1,344 (157) UBS Securities Australia Ltd Sell 1,111 (120) UBS Securities Australia Ltd Sell 2,391 (1314) UBS Securities Australia Ltd Sell 2,911 (142) UBS Securities Australia Ltd Sell 2,911 (142) UBS Securities Australia Ltd Sell 2,911 (142) UBS Securities Australia Ltd Sell 2,913 (142) UBS Securities Australia Ltd Sell 2,744 (170) UBS Securities Australia Ltd Sell 2,748 (800) UBS Securities Australia Ltd Sell 2,742 (1,65) UBS Securities Australia L	3-May-14	UBS Securities Australia Ltd	Sell	772	(83)	Ordinary
UBS Securities Australia Ltd Buy 1,004 108 UBS Securities Australia Ltd Sell 2,335,000 (57) UBS Securities Australia Ltd Sell 1,454 (150) UBS Securities Australia Ltd Sell 611 (65) UBS Securities Australia Ltd Sell 1,311 (100) UBS Securities Australia Ltd Sell 2,913 (142) UBS Securities Australia Ltd Sell 2,913 (103) UBS Securities Australia Ltd Sell 2,943 (103) UBS Securities Australia Ltd Sell 2,943 (103) UBS Securities Australia Ltd Sell 2,943 (103) UBS Securities Australia Ltd Sell 2,786 (300) UBS Securities Australia Ltd Sell 2,786 (300) UBS Securities Australia Ltd Sell 2,571 (107) UBS Securities Australia Ltd Sell 2,571 (107) UBS Securities Australia Ltd Sell 2,571 (107) UBS Securities Australia Ltd <td>3-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>1,618</td> <td>(174)</td> <td>Ordinary</td>	3-May-14	UBS Securities Australia Ltd	Sell	1,618	(174)	Ordinary
UBS Securities Australia Ltd Sell 2,325,000 (157) UBS Securities Australia Ltd Sell 6,20 (67) UBS Securities Australia Ltd Sell 1,111 (100) UBS Securities Australia Ltd Sell 9,033 (662) UBS Securities Australia Ltd Sell 9,731 (682) UBS Securities Australia Ltd Sell 9,731 (1420) UBS Securities Australia Ltd Sell 9,73 (105) UBS Securities Australia Ltd Sell 2,743 (1422) UBS Securities Australia Ltd Sell 2,786 (300) UBS Securities Australia	3-May-14	UBS Securities Australia Ltd	Buy	1,004	108	Ordinary
UBS Securities Australia Ltd Sell (1,57) UBS Securities Australia Ltd Sell (1,111 (120) UBS Securities Australia Ltd Sell (1,111 (120) UBS Securities Australia Ltd Sell 2,911 (165) UBS Securities Australia Ltd Sell 2,911 (105) UBS Securities Australia Ltd Sell 2,913 (105) UBS Securities Australia Ltd Sell 2,174 (105) UBS Securities Australia Ltd Sell 2,174 (105) UBS Securities Australia Ltd Sell 2,786 (100) UBS Securities Australia Ltd Sell 2,786 (100) UBS Securities Australia Ltd Sell 2,786 (100) UBS Securities Australia Ltd Sell 2,788 (107) UBS Securities Australia Ltd Sell 2,788 (107) UBS Securities Australia Ltd Sell 2,728 (107) UBS Securities Australia Ltd Sell 2,728 (107) UBS Securities Australia Ltd Se	3-May-14	UBS Securities Australia Ltd	Sell	2,325,000	(250,000)	Ordinary
UBS Securities Australia Ltd Sell 6.20 (67) UBS Securities Australia Ltd Sell 1.11 (120) UBS Securities Australia Ltd Sell 9,093 (682) UBS Securities Australia Ltd Sell 2,911 (314) UBS Securities Australia Ltd Sell 2,943 (105) UBS Securities Australia Ltd Sell 8,166 (879) UBS Securities Australia Ltd Sell 2,843 (105) UBS Securities Australia Ltd Sell 2,786 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 2,228 (730) UBS Securities Australia Ltd <td>3-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell Sell</td> <td>1,454</td> <td>(157)</td> <td>Ordinary</td>	3-May-14	UBS Securities Australia Ltd	Sell Sell	1,454	(157)	Ordinary
UBS Securities Australia Ltd Sell 6.111 (165) UBS Securities Australia Ltd Sell 6.903 (882) UBS Securities Australia Ltd Sell 2,311 (142) UBS Securities Australia Ltd Sell 8,166 (873) UBS Securities Australia Ltd Sell 2,174 (234) UBS Securities Australia Ltd Sell 2,174 (234) UBS Securities Australia Ltd Sell 2,786 (300) UBS Securities Australia Ltd Sell 2,786 (300) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 2,786 (300) UBS Securities Australia Ltd Sell 2,786 (300) UBS Securities Australia Ltd Sell 2,728 (264) UBS Securities Australia Ltd Sell 2,723 (107) UBS Securities Australia Ltd	3-May-14	UBS Securities Australia Ltd	Sell	079	(9)	Ordinary
UBS Securities Australia Ltd Sell 9.033 (982) UBS Securities Australia Ltd Sell 9.033 (982) UBS Securities Australia Ltd Sell 2.911 (314) UBS Securities Australia Ltd Sell 1.319 (412) UBS Securities Australia Ltd Sell 2.369 (254) UBS Securities Australia Ltd Sell 2.369 (255) UBS Securities Australia Ltd Sell 2.786 (300) UBS Securities Australia Ltd Sell 2.786 (300) UBS Securities Australia Ltd Sell 6.500 (700) UBS Securities Australia Ltd Sell 6.500 (700) UBS Securities Australia Ltd Sell 6.500 (700) UBS Securities Australia Ltd Sell 2.378 (30) UBS Securities Australia Ltd Sell 2.571 (406) UBS Securities Australia Ltd Sell 2.571 (406) UBS Securities Australia Ltd Sell 2.571 (406) UBS Securities Australia Ltd<	D-Ividy-14	ODS Securities Australia Ltd	Dell C-10	111,	(071)	Ordinary
UBS Securities Australia Ltd Sell 2,913 (192) UBS Securities Australia Ltd Sell 2,911 (314) UBS Securities Australia Ltd Sell 2,911 (105) UBS Securities Australia Ltd Sell 2,843 (306) UBS Securities Australia Ltd Sell 2,843 (306) UBS Securities Australia Ltd Sell 2,786 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 2,228 (1,807) UBS Securities Australia Ltd Sell 2,423 (1,607) UBS Securities Australia	3-May-14	UBS Securities Australia Ltd	J. Sell	119	(99)	Ordinary
UBS Securities Australia Ltd Sell 973 (105) UBS Securities Australia Ltd Sell 1,319 (142) UBS Securities Australia Ltd Sell 2,174 (224) UBS Securities Australia Ltd Sell 2,174 (224) UBS Securities Australia Ltd Sell 2,786 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 2,786 (800) UBS Securities Australia Ltd Sell 2,500 (700) UBS Securities Australia Ltd Sell 2,500 (700) UBS Securities Australia Ltd Sell 2,500 (700) UBS Securities Australia Ltd Sell 2,571 (107) UBS Securities Australia Ltd Sell 2,571 (107) UBS Securities Australia Ltd Sell 3,750 (107) UBS Securities Australia Ltd Sell 2,7423 (264) UBS Securities Australia Ltd<	3-May-14	IRS Securities Australia Ltd	Jac Jo	9,093	(382)	Ordinary
UBS Securities Australia Ltd Sell 1,319 (142) UBS Securities Australia Ltd Sell 1,319 (142) UBS Securities Australia Ltd Sell 2,843 (365) UBS Securities Australia Ltd Sell 2,174 (234) UBS Securities Australia Ltd Sell 2,769 (300) UBS Securities Australia Ltd Sell 2,786 (300) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 3,435 (700) UBS Securities Australia Ltd Sell 2,571 (270) UBS Securities Australia Ltd Sell 2,571 (406) UBS Securities Australia Ltd Sell 2,571 (406) UBS Securities Australia Ltd Sell 3,760 (406) UBS Securities Australia Ltd	May 14	Tibo Counties Australia Lita	100	116.0	(3.14)	Ordinary
UBS Securities Australia Ltd Sell 8.165 (879) UBS Securities Australia Ltd Sell 2.843 (366) UBS Securities Australia Ltd Sell 2.174 (253) UBS Securities Australia Ltd Sell 2.786 (700) UBS Securities Australia Ltd Sell 6.500 (700) UBS Securities Australia Ltd Sell 2.735 (107) UBS Securities Australia Ltd Sell 2.728 (107) UBS Securities Australia Ltd Sell 2.728 (107) UBS Securities Australia Ltd Sell 2.728 (107) UBS Securities Australia Ltd	3-May-14	I IRS Securities Australia Ltd	, A	1 210	(103)	Ordinary
UBS Securities Australia Ltd Sell 2,174 (624) UBS Securities Australia Ltd Sell 2,174 (234) UBS Securities Australia Ltd Sell 2,369 (730) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 3,560 (700) UBS Securities Australia Ltd Sell 2,38 (800) UBS Securities Australia Ltd Sell 2,35 (800) UBS Securities Australia Ltd Sell 2,57 (700) UBS Securities Australia Ltd Sell 2,57 (107) UBS Securities Australia Ltd Sell 2,57 (107) UBS Securities Australia Ltd Sell 2,73 (107) UBS Securities Australia Ltd Sell 3,70 (406) UBS Securities Australia Ltd Sell 3,70 (406) UBS Securities Australia Ltd Sell 3,70 (406) UBS Securities Australia Ltd	3-May-14	IRS Societies Australia Ltd	1100	391.0	(020)	Ordinary
UBS Securities Australia Ltd Sell 2,744 (234) UBS Securities Australia Ltd Sell 2,786 (200) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 7,428 (800) UBS Securities Australia Ltd Sell 7,358 (700) UBS Securities Australia Ltd Sell 1,536 (1,825) UBS Securities Australia Ltd Sell 1,536 (1,825) UBS Securities Australia Ltd Sell 2,528 (240) UBS Securities Australia Ltd Sell 2,717 (277) UBS Securities Australia Ltd Sell 3,709 (465) UBS Securities Australia Ltd Sell 3,700 (405) UBS Securities Australia Ltd Sell 3,700 (405) UBS Securities Australia Ltd Sell 3,700 (405) UBS Securities Australia	3-May-14	1 88 Societies Australia Ital	19	00,100	(906)	Ordinary
UBS Securities Australia Ltd Sell 2,174 (255) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 30,650 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 2,338 (370) UBS Securities Australia Ltd Sell 2,571 (1,825) UBS Securities Australia Ltd Sell 2,571 (405) UBS Securities Australia Ltd Sell 2,571 (405) UBS Securities Australia Ltd Sell 3,700 (405) UBS Securities Australia	May 14	U.D.C. Constraint Australia Lite	Jell Cell	2,043	(one)	Ordinary
UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 7,428 (800) UBS Securities Australia Ltd Sell 2,350 (700) UBS Securities Australia Ltd Sell 2,350 (700) UBS Securities Australia Ltd Sell 2,571 (1,825) UBS Securities Australia Ltd Sell 2,571 (1,825) UBS Securities Australia Ltd Sell 2,571 (1,07) UBS Securities Australia Ltd Sell 2,728 (2,60) UBS Securities Australia Ltd Sell 3,700 (406) UBS Securities Australia Ltd Sell 3,700 (406) UBS Securities Australia Ltd Sell 3,700 (406) UBS Securities Australia Ltd Sell 1,244 (134) UBS Securities Australia Ltd Sell 1,242 (150) UBS Securities Austral	3-May-14	I IRS Sociation Australia 1td	100	2,1/4	(254)	Ordinary
UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 7,825 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 1,6,536 (700) UBS Securities Australia Ltd Sell 1,6,336 (1,825) UBS Securities Australia Ltd Sell 1,6,336 (1,07) UBS Securities Australia Ltd Sell 2,771 (406) UBS Securities Australia Ltd Sell 3,770 (406) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,750 (405) UBS Securities Aust	3-May 14	I IBC Countries Australia I to	100	2,309	(002)	Ordinary
UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 30,550 (700) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 2,358 (700) UBS Securities Australia Ltd Sell 2,358 (700) UBS Securities Australia Ltd Sell 2,571 (107) UBS Securities Australia Ltd Sell 2,571 (107) UBS Securities Australia Ltd Sell 2,771 (405) UBS Securities Australia Ltd Sell 2,771 (405) UBS Securities Australia Ltd Sell 3,770 (405) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,750 (100) UBS Securities Australia Lt	New 2	I IBC Countries Australia Ltd	100	2000,0	(00/)	Oldinaly
UBS Securities Australia Ltd Sell 9,500 (VOD) UBS Securities Australia Ltd Sell 7,428 (800) UBS Securities Australia Ltd Sell 2,556 (700) UBS Securities Australia Ltd Sell 3,455 (800) UBS Securities Australia Ltd Sell 3,455 (1,825) UBS Securities Australia Ltd Sell 2,571 (1,825) UBS Securities Australia Ltd Sell 2,771 (405) UBS Securities Australia Ltd Sell 3,700 (406) UBS Securities Australia Ltd Sell 1,244 (134) UBS Securities Australia Ltd Sell 1,244 (134) UBS Securities Australia Ltd Sell 1,546 (105) UBS Securities Australia Ltd Sell 2,759 (100) UBS Securities Australia	2 May 14	LIBC Counties Australia Ltd	lac l	7,780	(300)	Orginary
UBS Securities Australia Ltd Sell 7,020 (3,507) UBS Securities Australia Ltd Sell 6,500 (700) UBS Securities Australia Ltd Sell 1,535 (700) UBS Securities Australia Ltd Sell 1,6336 (1,823) UBS Securities Australia Ltd Sell 1,6336 (1,823) UBS Securities Australia Ltd Sell 2,571 (240) UBS Securities Australia Ltd Sell 2,573 (405) UBS Securities Australia Ltd Sell 2,423 (405) UBS Securities Australia Ltd Sell 3,700 (405) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,750 (405) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,759 (1,061) UBS Securities	May 14	Upo Counties Australia Ltd	Sell	00000	(00/)	Orginary
UBS Securities Australia Ltd Sell 6,725 UBS Securities Australia Ltd Sell 2,358 (254) UBS Securities Australia Ltd Sell 16,936 (1070) UBS Securities Australia Ltd Sell 16,936 (1077) UBS Securities Australia Ltd Sell 2,571 (240) UBS Securities Australia Ltd Sell 2,571 (406) UBS Securities Australia Ltd Sell 3,700 (406) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,759 (1,00) UBS Securities Australia Ltd Sell 9,759 (1,00) UBS Securities Australia Ltd <	3-May-14	UBS Securities Australia Hd	Soll	00,000	(3,301)	Ordinary
UBS Securities Australia Ltd Sell 2,556 (754) UBS Securities Australia Ltd Sell 3,455 (324) UBS Securities Australia Ltd Sell 16,936 (1,825) UBS Securities Australia Ltd Sell 2,571 (1,825) UBS Securities Australia Ltd Sell 2,571 (406) UBS Securities Australia Ltd Sell 3,700 (406) UBS Securities Australia Ltd Sell 3,700 (406) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 1,244 (134) UBS Securities Australia Ltd Sell 9,759 (1051) UBS Securities Australia Ltd Sell 9,759 (1051) UBS Securities Australia Ltd Sell 9,759 (100,000) UBS Securities Australia Ltd Sell 2,900 (100,000) UBS Securities Australia Ltd Sell 2,900 (100,000) UBS Secur	3-May-14	I IBS Societies Australia 14d	il o	074'/	(002)	Ordinary
UBS Securities Australia Ltd Sell 1,536 (1,329) UBS Securities Australia Ltd Sell 16,936 (1,825) UBS Securities Australia Ltd Sell 16,936 (1,825) UBS Securities Australia Ltd Sell 2,527 (240) UBS Securities Australia Ltd Sell 3,70 (406) UBS Securities Australia Ltd Sell 3,70 (406) UBS Securities Australia Ltd Sell 2,423 (240) UBS Securities Australia Ltd Sell 2,423 (406) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securi	3-May-14	I IBA Con initias Australia ++4	Coll	00,000	(700)	Ordinary
UBS Securities Australia Ltd Sell 1,370 UBS Securities Australia Ltd Sell 16,936 (107) UBS Securities Australia Ltd Sell 2,571 (207) UBS Securities Australia Ltd Sell 3,770 (406) UBS Securities Australia Ltd Sell 3,770 (406) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 9,759 (1,031) UBS Securities Australia Ltd Sell 9,759 (1,031) UBS Securities Australia Ltd Sell 9,759 (1,000) UBS Securities Australia Ltd Sell 9,750 (100,000) UBS Securities Australia Ltd Sell 9,750 (100,000) UBS Securities Australia Ltd	2 May 14	UBS Securities Australia Ltd	Sell	2,538	(254)	Ordinary
UBS Securities Australia Lud Sell 1,0,209 1,0,201 1,0,209 1,0,209 1,0,209 1,0,209 1,0,209 1,0,209 1,0,209 1,0,209 1,0,209 1,0,209 1,0,209 1,0,209 1,0,201 1,0,2	3-May-14	U.B.C. Contraction Australia 1+4	Sell	2,433	(370)	Ordinary
UBS Securities Australia Ltd Sell 2,571 (177) UBS Securities Australia Ltd Sell 2,228 (240) UBS Securities Australia Ltd Sell 3,770 (406) UBS Securities Australia Ltd Sell 2,423 (261) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 9,789 (43) UBS Securities Australia Ltd Sell 9,789 (1,631) UBS Securities Australia Ltd Sell 9,789 (1,631) UBS Securities Australia Ltd Sell 9,780 (1,601) UBS Securities Australia Ltd Sell 2,99,000 (1,600,000) UBS Securities Australia Ltd Sell 2,787,000 (100,000)	3-May-14	URS Securities Australia I to	Sell	000	(201)	Ordinary
UBS Securities Australia Ltd Sell 2,228 (240) UBS Securities Australia Ltd Sell 3,770 (406) UBS Securities Australia Ltd Sell 501 (54) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,750 (405) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,29,000 (100,000) UBS Securities Australia Ltd Sell 2,781,000 (100,000) UBS Securities Australia Ltd Sell 2,786,000 (300,000)	3-May-14	UBS Securities Australia I td	3	177.7	(101)	Ordinary
UBS Securities Australia Ltd Sell 3,770 (466) UBS Securities Australia Ltd Sell 5.01 (54) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 3,760 (405) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 9,29,000 (10,000) UBS Securities Australia Ltd Sell 2,79,700 (10,000) UBS Securities Australia Ltd Sell 2,750 (10,000) <t< td=""><td>3-Mav-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>2,228</td><td>(240)</td><td>Ordinary</td></t<>	3-Mav-14	UBS Securities Australia Ltd	Sell	2,228	(240)	Ordinary
UBS Securities Australia Ltd Sell 501 (54) UBS Securities Australia Ltd Sell 2423 (261) UBS Securities Australia Ltd Sell 3.760 (405) UBS Securities Australia Ltd Sell 1,244 (134) UBS Securities Australia Ltd Sell 9,759 (403) UBS Securities Australia Ltd Sell 9,759 (150) UBS Securities Australia Ltd Sell 1,392 (150) UBS Securities Australia Ltd Sell 9,29,000 (100,000) UBS Securities Australia Ltd Sell 2,29,000 (100,000) UBS Securities Australia Ltd Sell 2,787,000 (100,000) UBS Securities Australia Ltd Sell 2,787,000 (100,000) UBS Securities Australia Ltd Sell 2,787,000 (300,000) UBS Securities Australia Ltd Sell 2,787,000 (300,000) UBS Securities Australia Ltd Sell 2,787,000 (300,000) UBS Securities Australia Ltd Sell (300,000) (300,000)	3-Mav-14	UBS Securities Australia Ltd	S	3 770	(406)	Ordinary
UBS Securities Australia Ltd Self 2,423 (261) UBS Securities Australia Ltd Self 3,760 (405) UBS Securities Australia Ltd Self 1,244 (134) UBS Securities Australia Ltd Self 9,759 (4,051) UBS Securities Australia Ltd Self 9,759 (1,051) UBS Securities Australia Ltd Self 9,900 (10,000) UBS Securities Australia Ltd Self 929,000 (10,000) UBS Securities Australia Ltd Self 929,000 (10,000) UBS Securities Australia Ltd Self 2,785,000 (10,000) UBS Securities Australia Ltd Self 2,785,000 (300,000) UBS Securities Australia Ltd Self 2,785,000 (300,000) UBS Securities Australia Ltd Self 5,785,000 (300,000) UBS Securities Australia Ltd Self 5,785,000 (300,000)	3-May-14	UBS Securities Australia Ltd	Sell	501	(54)	Ordinary
UBS Securities Australia Ltd Self 3,760 (405) UBS Securities Australia Ltd Self 1,244 (134) UBS Securities Australia Ltd Self 359 (4,33) UBS Securities Australia Ltd Self 9,759 (1,051) UBS Securities Australia Ltd Self 6,165 (664) UBS Securities Australia Ltd Self 9,759 (1,051) UBS Securities Australia Ltd Self 9,759 (1,001) UBS Securities Australia Ltd Self 9,750 (100,000) UBS Securities Australia Ltd Self 2,79,000 (100,000) UBS Securities Australia Ltd Self 2,79,000 (100,000) UBS Securities Australia Ltd Self 2,79,000 (100,000) UBS Securities Australia Ltd Self 5,759 5,759	3-May-14	UBS Securities Australia Ltd	Sell	2.423	(261)	Ordinary
UBS Securities Australia Ltd Sell 1,244 (134) UBS Securities Australia Ltd Sell 359 (43) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 6,165 (664) UBS Securities Australia Ltd Sell 1,392 (150) UBS Securities Australia Ltd Sell 929,000 (100,000) UBS Securities Australia Ltd Sell 2787,000 (100,000) UBS Securities Australia Ltd Buy 48,915 5,719 UBS Securities Australia Ltd Sell 28,7109 (100,000) UBS Securities Australia Ltd Buy 48,915 5,719 UBS Securities Australia Ltd Sell (73,055) 5,719	3-May-14	UBS Securities Australia Ltd	Self	3,760	(405)	Ordinary
UBS Securities Australia Ltd Sell 9399 (43) UBS Securities Australia Ltd Sell 9,759 (1,051) UBS Securities Australia Ltd Sell 6,165 (164) UBS Securities Australia Ltd Sell 92,000 (1) UBS Securities Australia Ltd Sell 92,000 (100,000) UBS Securities Australia Ltd Sell 2,787,000 (300,000) UBS Securities Australia Ltd Buy 48,915 5,219 UBS Securities Australia Ltd Sell 5,787,000 (300,000) UBS Securities Australia Ltd Sell 5,787,000 (300,000) UBS Securities Australia Ltd Sell 5,787,000 (300,000)	3-May-14	UBS Securities Australia Ltd	Sell	1,244	(134)	Ordinary
UBS Securities Australia Ltd Sell 6,155 (1,051) UBS Securities Australia Ltd Sell 1,332 (150) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 929,000 (100,000) UBS Securities Australia Ltd Sell 22,787,000 (300,000) UBS Securities Australia Ltd Buy 48,915 5,000 UBS Securities Australia Ltd Sell 2,787,000 (300,000) UBS Securities Australia Ltd Sell 8,000 (300,000)	3-May-14	UBS Securities Australia Ltd	Self	399	(43)	Ordinary
UBS Securities Australia Ltd Sell 6,165 (664) UBS Securities Australia Ltd Sell 1,392 (150) UBS Securities Australia Ltd Sell 9,29,000 (10,000) UBS Securities Australia Ltd Sell 2,29,000 (100,000) UBS Securities Australia Ltd Sell 2,79,000 (300,000) UBS Securities Australia Ltd Buy 48,915 5,219 UBS Securities Australia Ltd Sell (73,055) 5,219	3-May-14	UBS Securities Australia Ltd	Sell	652'6	(1,051)	Ordinary
UBS Securities Australia Ltd Sell 1,392 (150) UBS Securities Australia Ltd Sell 9 (1) UBS Securities Australia Ltd Sell 2,787,000 (100,000) UBS Securities Australia Ltd Buy 48,915 5,219 UBS Securities Australia Ltd Sell 687,096 (73,065)	3-May-14	UBS Securities Australia Ltd	Sell	6,165	(664)	Ordinary
UBS Securities Australia Ltd Sell 99 (1) UBS Securities Australia Ltd Sell 2,767,000 (300,000) UBS Securities Australia Ltd Buy 48,915 5,219 UBS Securities Australia Ltd Sell 2,787,000 (300,000) UBS Securities Australia Ltd Buy 48,915 5,219 UBS Securities Australia Ltd Sell 687,096 (73,065)	3-May-14	UBS Securities Australia Ltd	Sell	1,392	(120)	Ordinary
UBS Securities Australia Ltd Sell 9229,000 (100,000) UBS Securities Australia Ltd Sell 2,785,000 (300,000) UBS Securities Australia Ltd Buy 48,915 5,219 UBS Securities Australia Ltd Sell 687,096 (73,065)	3-May-14	UBS Securities Australia Ltd	Sell	6	(E)	Ordinary
UBS Securities Australia Ltd Sell 2,787,000 300,000) UBS Securities Australia Ltd Buy 48,915 5,219 UBS Securities Australia Ltd Sell 68,70,96 (73,065)	3-May-14	UBS Securities Australia Ltd	Sell	929,000	(100,000)	Ordinary
UBS Securities Australia Ltd Buy 48,915 5,219 UBS Securities Australia Ltd Sell 687,096 (73,065)	3-May-14	UBS Securities Australia Ltd	Sell	2,787,000	(300,000)	Ordinary
UBS Securities Australia Ltd Sell (73,065)	-May-14	UBS Securities Australia Ltd	Buy	48,915	5,219	Ordinary
	5-May-14					

23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	A/N	1,219	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NVA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝΑ	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑΛ	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NVA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΑΝ	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA VA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	Wealth Management	Transfer-In	NA VA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	A.N.	121	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝA	1,829	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝΑ	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝΑ	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,829	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	609	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	121	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	121	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝΑ	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	121	Ordinary
-May-14	UBS Weaith Management Australia Ltd	Transfer-in	NA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-in	ΑN	121	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-in	NA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	609	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝA	1,219	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	₩A	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	ΝΑ	609	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	121	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	121	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	609	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,219	Ordinary
3-May-14	UBS Wealth Management Australia Ltd	Transfer-In	N/A	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-in	NVA	1,829	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Transfer-in	N/A	1,219	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	NA	1,829	Ordinary
23-May-14	UBS Wealth Management Australia Ltd	Transfer-In	WA	1,829	Ordinary
73-May-14					
	II IN Wealth Management Australia Ltd	Transfer-In	AN.	1.829	Ordinary

BS Securities Authorise Let Self	11-11	1 IDS Cocypition Appetralia 14d	= 5			The same of the same of
Bits Securities Autorialis Ltd Self Se	lav-14	UBS Securities Australia I to	R 3	2 822	(300)	Ordinary
Big Securities Autorials Ltd Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67) (6.67) Big Securities Autorials Ltd Sell Sell 6.973 (6.67)	lay-14	UBS Securities Australia Ltd	Sell	9.405	(1,000)	Ordinary
Bass Securities Autoratival Ltd Seal S	lay-14	UBS Securities Australia Ltd	Sell	2,173	(231)	Ordinary
BES Securities Autoriale Ltd Sell 9.533 (1507)	26-May-14	UBS Securities Australia Ltd	Sell	4,392	(467)	Ordinary
UBS Securities Autoration Ltd Sell 1,9,00 (1,00) UBS Securities Autoration Ltd Sell 1,6,10 (1,54,0) UBS Securities Autoration Ltd Sell 5,00 (1,01) UBS Securities Autoration Ltd Sell 2,00 (1,01) UBS Securities Autoration Ltd Sell 2,47 (1,54,0) UBS Securities Autoration Ltd Sell 2,47 (1,64,0) UBS Securities Autoration Ltd Sell 2,560 (1,01) UBS Securities Autoration Ltd Sell 2,50 (1,01) UBS Securities Autoration Ltd Sell 2,50 (1,01) UBS Securities Autoration Ltd Sell 2,50 (1,01) UBS Securities Autoration Ltd Sell 3,00 (2,00) UB	Aay-14	UBS Securities Australia Ltd	ll-SC	5,333	(267)	Ordinary
Bigs Securities Australia Ltd Sell 11,091	/ay-14	UBS Securities Australia Ltd	Sell	9,405	(1,000)	Ordinary
UBS Securities Australia Ltd Sell 14,514	Aay-14	UBS Securities Australia Ltd	Seil	1,091	(116)	Ordinary
USS Securities Australia Ltd Sell Sell 44,72 (476) USS Securities Australia Ltd Sell Sell 44,72 (476) USS Securities Australia Ltd Sell Sell 15,707 (1,671) USS Securities Australia Ltd Sell Sell 15,707 (1,671) USS Securities Australia Ltd Sell Sell 1,5707 (1,671) USS Securities Australia Ltd Sell Sell 1,5707 (1,671) USS Securities Australia Ltd Sell Sell Sell 1,5707 (1,671) USS Securities Australia Ltd Sell Sell Sell 1,5707 (1,671) USS Securities Australia Ltd Sell Sell Sell 1,5707 (1,671) USS Securities Australia Ltd Sell	/ay-14	UBS Securities Australia Ltd	Seil	14,514	(1,544)	Ordinary
UBS Securities Australia Ltd Sell 4,474 (476) (476) UBS Securities Australia Ltd Sell 24,120 (2,566) UBS Securities Australia Ltd Sell 24,120 (2,566) UBS Securities Australia Ltd Sell Sell 24,120 (2,566) UBS Securities Australia Ltd Sell Se	- 1	UBS Securities Australia Ltd	Seil	5,076	(240)	Ordinary
USS Securities Australia Ltd Sell Sell 24,100 10,100 USS Securities Australia Ltd Sell Sell 24,100 10,260 USS Securities Australia Ltd Sell Sell 15,000 10,200 USS Securities Australia Ltd Sell Sell 13,000 13,000 USS Securities Australia Ltd Sell Sell 14,300 13,000 USS Securities Australia Ltd Sell Sell Sell 14,300 13,000 USS Securities Australia Ltd Sell Sell Sell 14,300 10,000 USS Securities Australia Ltd Sell Sell Sell 14,000 10,000 USS Securities Australia Ltd Sell Sell Sell 1,000 10,000 USS Securities Australia Ltd Sell Sell Sell 1,000 1,000 1,000 USS Securities Australia Ltd Sell Sell Sell 1,000 1,000 1,000 1,000 USS Securities Australia Ltd Sell Sell Sell 2,000 1,000 1,000 USS Securities Australia Ltd Sell Sell 2,000 1,000 1,000 USS Securities Australia Ltd Sell Sell 2,000 1,000 1,000 1,000 USS Securities Australia Ltd Sell Sell 2,000 1,000 1,000 1,000 USS Securities Australia Ltd Sell Sell 2,000 1	- 1	UBS Securities Australia Ltd	Sell	4,474	(476)	Ordinary
UBS Securities Autorials Ltd Sell 15,170 (1,671) UBS Securities Autorials Ltd Sell 15,022 (260) UBS Securities Autorials Ltd Sell 1,622 (280) UBS Securities Autorials Ltd Sell 3,008 (320) UBS Securities Autorials Ltd Sell 3,008 (320) UBS Securities Autorials Ltd Sell 3,008 (320) UBS Securities Autorials Ltd Sell 14,705 (1,671) UBS Securities Autorials Ltd Sell 14,705 (1,671) UBS Securities Autorials Ltd Sell 4,705 (500) UBS Securities Autorials Ltd Sell 4,705 (500) UBS Securities Autorials Ltd Sell 4,705 (500) UBS Securities Autorials Ltd Buy 3,100 3,100 UBS Securities Autoria	- 1	UBS Securities Australia Ltd	Sell	950	(101)	Ordinary
Bits Securities Australia Ltd Sell 15,707 (1,57) UBS Securities Australia Ltd Sell 8,50 (951) UBS Securities Australia Ltd Sell 8,50 (950) UBS Securities Australia Ltd Sell 3,50 (220) UBS Securities Australia Ltd Sell 8,50 (320) UBS Securities Australia Ltd Sell 8,50 (300) UBS Securities Australia Ltd Sell 8,50 (300) UBS Securities Australia Ltd Sell 4,70 (300) UBS Securities Australia Ltd Buy 3,51 3,50 UBS Securities Australia Ltd Buy 3,51 3,60 UBS Securities Australia Ltd Buy 3,51 3,60 UBS Securities Australia Ltd Buy 3,51 3,60 UBS Securities Australia Ltd B	- 1	UBS Securities Australia Ltd	Sell	24,120	(2,566)	Ordinary
BBS Securities Australia Ltd Sell 8.939 (951) BBS Securities Australia Ltd Sell 3.000 (2.92) BBS Securities Australia Ltd Sell 3.000 (3.20) BBS Securities Australia Ltd Sell 3.000 (3.20) BBS Securities Australia Ltd Sell 3.000 (3.20) BBS Securities Australia Ltd Sell 4.705,000 (3.20) (3.20) BBS Securities Australia Ltd Sell 4.705,000 (3.20) (3.2	May-14	UBS Securities Australia Ltd	Sell	15,707	(1,671)	Ordinary
UBS Securities Australia Ltd Sell 2,572 (280) UBS Securities Australia Ltd Sell 2,572 (280) UBS Securities Australia Ltd Sell 35,506 (377) UBS Securities Australia Ltd Sell 35,507 (377) UBS Securities Australia Ltd Sell 4,765 (307) UBS Securities Australia Ltd Sell 4,765 (307) UBS Securities Australia Ltd Sell 4,765 (307) UBS Securities Australia Ltd Sell Buy 2,592 (37) (2,500) UBS Securities Australia Ltd Sell Buy 3,572 (37) (2,500) UBS Securities Australia Ltd Sell Buy 3,572 (37) (2,500) UBS Securities Australia Ltd Sell Buy 3,572 (37) (2,500) UBS Securities Australia Ltd Sell Sell 4,705,000 (500,000) UBS Securities Australia Ltd Sell Sell 3,572 (500,000) Sell 4,705,000 (500,000) UBS Securities Australia Ltd Sell Sell 3,572 (500,000) Sell 4,705,000 (500,000) UBS Securities Australia Ltd Sell Sell 3,572 (500,000) Sell 3,572 (500,00	May-14	UBS Securities Australia Ltd	Sell	8,939	(951)	Ordinary
May Securities Australia Ltd Sell 3,055 (220) May Securities Australia Ltd Sell 3,050 (3,777) May Securities Australia Ltd Sell 3,500 (3,777) May Securities Australia Ltd Sell 4,705,000 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd Sell 7,900 (5,00,000) May Securities Australia Ltd 5,900 (5,00,000)		UBS Securities Australia Ltd	Sell	82	(6)	Ordinary
UBS Securities Australia Ltd Sell 35,008 (3777) UBS Securities Australia Ltd Sell 35,008 (3777) UBS Securities Australia Ltd Sell 4,438 (1,530) UBS Securities Australia Ltd Sell 4,766 (507) UBS Securities Australia Ltd Sell 4,766 (507) UBS Securities Australia Ltd Sell 4,705,000 (50,000) UBS Securities Australia Ltd Sell 4,705,000 (50,000) UBS Securities Australia Ltd Buy 2,707,500 (50,000) UBS Securities Australia Ltd Buy 3,170 3,207 (3,100) UBS Securities Australia Ltd Buy 3,170 3,207 (3,100) UBS Securities Australia Ltd Buy 3,170 3,207 (3,100) UBS Securities Australia Ltd Buy 3,170 3,20 (3,100) UBS Securities Australia Ltd Buy 3,170 3,20 (3,100) UBS Securities Australia Ltd Buy 3,170 (3,200) UBS Securities A	- 1	UBS Securities Australia Ltd	Sell	2,632	(280)	Ordinary
UBS Securities Australia Ltd Sell 81,504 (8,37) UBS Securities Australia Ltd Sell 8,404 (8,90) UBS Securities Australia Ltd Sell 4,766 (50) UBS Securities Australia Ltd Sell 4,766 (50) UBS Securities Australia Ltd Sell 4,766 (62) UBS Securities Australia Ltd Sell 4,766 (62) UBS Securities Australia Ltd Buy 2,927 (142) UBS Securities Australia Ltd Buy 2,927 (150) UBS Securities Australia Ltd Buy 3,105 331 UBS Securities Australia Ltd Buy 3,105 338 UBS Securities Australia Ltd Buy 3,105 3,105 UBS Securities Australia Ltd	- 1	UBS Securities Australia Ltd	Sell	3,008	(320)	Ordinary
UBS Securities Australia Ltd Sell 8,404 (8,90) UBS Securities Australia Ltd Sell 14,362 (507) UBS Securities Australia Ltd Sell 4,766 (507) UBS Securities Australia Ltd Sell 2,2616 (540) UBS Securities Australia Ltd Sell 2,2616 (540) UBS Securities Australia Ltd Sell 4,705,000 (500,000) UBS Securities Australia Ltd Buy 2,992 3,193 UBS Securities Australia Ltd Buy 3,107 3,00 UBS Securities Australia Ltd Buy 3,107 3,10 UBS Securities Australia Ltd Buy 3,107 3,10 UBS Securities Australia Ltd Buy 3,10 3,38 UBS Securities Australia Ltd Buy 3,10 3,38 UBS Securities Australia Ltd Buy 3,10 3,38 UBS Securities Australia Ltd Buy 3,10 3,50 UBS Securities Australia Ltd Sell 2,202,33 1,109 UBS Securities Australia Ltd <td>- 1</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>35,504</td> <td>(3,777)</td> <td>Ordinary</td>	- 1	UBS Securities Australia Ltd	Sell	35,504	(3,777)	Ordinary
UBS Securities Australia Ltd Seli 4,382 (1,530) UBS Securities Australia Ltd Seli 4,382 (1,501) UBS Securities Australia Ltd Seli 4,055 (29) UBS Securities Australia Ltd Seli 4,705 (1,501) UBS Securities Australia Ltd Seli 4,705 (500) UBS Securities Australia Ltd Buy 2,957 2,199 UBS Securities Australia Ltd Buy 2,0627 3,195 UBS Securities Australia Ltd Buy 2,0627 3,191 UBS Securities Australia Ltd Buy 3,105 3,105 UBS Securities Australia Ltd Buy 3,105 3,105 UBS Securities Australia Ltd Buy 3,100 3,100 UBS Securities Australia Ltd Buy 3,100 3,100 UBS Securities Australia Ltd Buy 4,650 5,00 UBS Securities Australia Ltd Buy 3,100 3,100 UBS Securities Australia Ltd Seli 2,37,60 5,00 UBS Securities Australia Ltd </td <td>- 1</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>8,404</td> <td>(894)</td> <td>Ordinary</td>	- 1	UBS Securities Australia Ltd	Sell	8,404	(894)	Ordinary
UBS Securities Australia Ltd Sell 4,766 (507) UBS Securities Australia Ltd Sell 2,2616 (421) UBS Securities Australia Ltd Sell 2,2616 (4,21) UBS Securities Australia Ltd Sell 4,705,000 (500,000) UBS Securities Australia Ltd Buy 2,992 319 UBS Securities Australia Ltd Buy 2,992 319 UBS Securities Australia Ltd Buy 2,661 3,712 609 UBS Securities Australia Ltd Buy 3,170 338 UBS Securities Australia Ltd Buy 3,170 3,88 UBS Securities Australia Ltd Sell 3,270 (1,129) UBS Securities Australia Ltd Sell 3,270 (1,500) UBS Securiti	May-14	UBS Securities Australia Ltd	Sell	14,382	(1,530)	Ordinary
UBS Securities Australia Ltd Sell 55 (9) UBS Securities Australia Ltd Sell 22,516 (2,406) UBS Securities Australia Ltd Sell 22,516 (2,406) UBS Securities Australia Ltd Buy 2,957 2,199 UBS Securities Australia Ltd Buy 2,957 3,195 UBS Securities Australia Ltd Buy 2,957 3,195 UBS Securities Australia Ltd Buy 2,957 3,195 UBS Securities Australia Ltd Buy 2,551 2,199 UBS Securities Australia Ltd Sell 2,340 3,00 UBS Securities Australia Ltd Sell 2,340 3,00 UBS Securities Australia Ltd Sell 2,340 3,00 UBS Securities Australia Ltd Sell	May-14	UBS Securities Australia Ltd	Self	4,766	(202)	Ordinary
UBS Securities Australia Ltd Sell 22,616 (421) UBS Securities Australia Ltd Sell 3,557 (421) UBS Securities Australia Ltd Sell 3,557 (421) UBS Securities Australia Ltd Buy 2,652 2,199 UBS Securities Australia Ltd Buy 2,652 3,19 UBS Securities Australia Ltd Buy 3,105 331 UBS Securities Australia Ltd Buy 3,105 338 UBS Securities Australia Ltd Buy 3,105 338 UBS Securities Australia Ltd Buy 3,105 3,88 UBS Securities Australia Ltd Sell 2,204,978 (2,500) UBS Securities Australia Ltd Sell 2,204,978 (2,140) UBS Securities Australia Ltd Sell 2,204,978 (2,140) UBS Securities Australia Ltd<	- 1	UBS Securities Australia Ltd	Sell	82	(6)	Ordinary
UBS Securities Australia Ltd Sell 2.2,616 UBS Securities Australia Ltd Sell 4,705,000 (2,409) UBS Securities Australia Ltd Buy 2,992 2,119 UBS Securities Australia Ltd Buy 2,992 2,119 UBS Securities Australia Ltd Buy 2,992 2,119 UBS Securities Australia Ltd Buy 3,150 3,172 UBS Securities Australia Ltd Buy 3,172 368 UBS Securities Australia Ltd Buy 3,170 3,18 UBS Securities Australia Ltd Buy 3,170 3,170 UBS Securities Australia Ltd Buy 3,170 3,170 UBS Securities Australia Ltd Buy 3,170 3,171 UBS Securities Australia Ltd Buy 3,20,20 (5,1,149) UBS Securities Australia Ltd Buy	- 1	UBS Securities Australia Ltd	J. Kell	208	(54)	Ordinary
UBS Securities Australia Ltd Sell 4,705,000 (50,000) UBS Securities Australia Ltd Buy 2,957 2,199 UBS Securities Australia Ltd Buy 2,0527 2,199 UBS Securities Australia Ltd Buy 2,0527 2,199 UBS Securities Australia Ltd Buy 2,957 2,199 UBS Securities Australia Ltd Buy 2,521 2,72 UBS Securities Australia Ltd Buy 3,170 3,88 UBS Securities Australia Ltd Sell 2,27,00 (1,129) UBS Securities Australia Ltd Sell 2,37,00 (1,129) UBS Securities Australia Ltd Sell 2,37,30 (1,129) UBS Securities Australia Ltd Sell 2,37,30 (1,129) UBS Securities Australia Ltd Sell 2,37,30 (1,129) UBS Securitie	- 1	UBS Securities Australia Ltd	le le	919,77	(2,406)	Ordinary
UBS Securities Australia Ltd Sell 4,707,500 (500,000) UBS Securities Australia Ltd Buy 2,922 319 UBS Securities Australia Ltd Buy 3,105 331 UBS Securities Australia Ltd Buy 3,105 338 UBS Securities Australia Ltd Buy 3,105 388 UBS Securities Australia Ltd Buy 3,105 3,00 UBS Securities Australia Ltd Buy 3,100 3,00 UBS Securities Australia Ltd Sell 1,0572 1,139 UBS Securities Australia Ltd Sell 2,304,978 2,81,149 UBS Securities Australia Ltd Sell 2,130 1,139 UBS Securities Australia Ltd Sell 2,130 1,139 UBS Securities Australia Ltd		Obsider Australia Lid	Sell	3,957	(421)	Ordinary
UBS Securities Australia Ltd Buy 2,927 2199 UBS Securities Australia Ltd Buy 2,927 2,139 UBS Securities Australia Ltd Buy 2,927 2,139 UBS Securities Australia Ltd Buy 3,452 3.88 UBS Securities Australia Ltd Buy 3,452 3.88 UBS Securities Australia Ltd Buy 3,170 3.88 UBS Securities Australia Ltd Buy 3,170 3.88 UBS Securities Australia Ltd Buy 3,170 3.88 UBS Securities Australia Ltd Buy 4,690 5.00 UBS Securities Australia Ltd Sell 10,672 (1,129) UBS Securities Australia Ltd Sell 23,700 (1,149) UBS Securities Australia Ltd Buy 2,304,978 (2,1,49) UBS Securities Australia Ltd Sell 2,304,978 (2,1,49) UBS Securities Australia Ltd Buy 2,304,978 (2,1,49) UBS Securities Australia Ltd Sell 2,304,978 (2,1,49) UBS Securiti	ı	Obsider Australia Ltd	Je J	4,705,000	(200,000)	Ordinary
UBS Securities Australia Ltd Buy 2,927 319 UBS Securities Australia Ltd Buy 2,927 3,195 UBS Securities Australia Ltd Buy 2,551 609 UBS Securities Australia Ltd Buy 3,105 338 UBS Securities Australia Ltd Buy 3,105 338 UBS Securities Australia Ltd Buy 3,100 3,80 UBS Securities Australia Ltd Buy 3,100 3,80 UBS Securities Australia Ltd Buy 3,80 4 UBS Securities Australia Ltd Sell 10,672 (1,129) UBS Securities Australia Ltd Sell 10,672 (1,129) UBS Securities Australia Ltd Sell 20,595 (1,149) UBS Securities Australia Ltd Sell 20,500 (1,129) UBS Securities Australia Ltd Sell 20,500 (1,149) UBS Securities Australia Ltd Sell 20,500 (1,150) UBS Securities Australia Ltd Sell 20,495 (1,150) UBS Securities Australia Ltd </td <td>ı</td> <td>UBS Securities Australia Ltd</td> <td>Neil Paris</td> <td>4,707,500</td> <td>(1000,000)</td> <td>Ordinary</td>	ı	UBS Securities Australia Ltd	Neil Paris	4,707,500	(1000,000)	Ordinary
UBS Securities Australia Ltd Buy 5,712 6.09 UBS Securities Australia Ltd Buy 3,105 331 UBS Securities Australia Ltd Buy 3,105 338 UBS Securities Australia Ltd Buy 2,551 272 UBS Securities Australia Ltd Buy 3,170 338 UBS Securities Australia Ltd Buy 3,170 338 UBS Securities Australia Ltd Buy 3,170 338 UBS Securities Australia Ltd Buy 3,170 3,18 4,140 UBS Securities Australia Ltd Sell 23,700 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,137 (1,149) 10,130 10,130 10,130 10,130 10,130 10,130 10,130 10,130 10,130 10,130 10,144 10,144 10,144	1	Obsideration Australia Ltd	buy Duty	7,007	2,199	Ordinary
UBS Securities Australia Ltd Buy 3,105 309 UBS Securities Australia Ltd Buy 3,105 338 UBS Securities Australia Ltd Buy 3,170 388 UBS Securities Australia Ltd Buy 4,690 500 UBS Securities Australia Ltd Buy 4,690 500 UBS Securities Australia Ltd Buy 4,690 500 UBS Securities Australia Ltd Sell 23,700 (1,129) UBS Securities Australia Ltd Sell 23,700 (1,149) UBS Securities Australia Ltd Sell 79,455 (8,14) UBS Securities Australia Ltd Sell 79,455 (8,14) UBS Securities Australia Ltd Sell 2,304,978 (5,1,149) UBS Securities Australia Ltd Sell 2,304,978 (6,1,149) UBS Securities Australia Ltd Sell 2,304,978 (6,1,34) UBS Securities Australia Ltd Sell 2,304,978 (6,1,34) UBS Securities Australia Ltd Sell 2,304,978 (6,1,39) <th< td=""><td>1</td><td>URC Congrise Australia 144</td><td>Bug</td><td>266,2</td><td>200</td><td>Ordinary</td></th<>	1	URC Congrise Australia 144	Bug	266,2	200	Ordinary
UBS Securities Australia Ltd Buy 3.51.02 381 UBS Securities Australia Ltd Buy 2.551 272 UBS Securities Australia Ltd Buy 2.551 272 UBS Securities Australia Ltd Buy 4.690 500 UBS Securities Australia Ltd Buy 4.690 500 UBS Securities Australia Ltd Stock bornowed NA 90,951 UBS Securities Australia Ltd Sell 23,700 (2,500) UBS Securities Australia Ltd Sell 20,269 (2,149) UBS Securities Australia Ltd Sell 20,269 (2,146) UBS Securities Australia Ltd Sell 2,304,978 (2,500) UBS Securities Australia Ltd Sell 2,502,169 (2,500)		Obsider Australia Ltd	, ng	2,712	500	Orginary
UBS Securities Australia Ltd Buy 2,551 272 UBS Securities Australia Ltd Buy 2,551 272 UBS Securities Australia Ltd Buy 3,170 338 UBS Securities Australia Ltd Buy 3,170 3,88 4 UBS Securities Australia Ltd Sell 10,672 (1,129) UBS Securities Australia Ltd Sell 23,700 (2,500) UBS Securities Australia Ltd Sell 20,737 (1,129) UBS Securities Australia Ltd Sell 20,737 (1,149) UBS Securities Australia Ltd Sell 20,458 (2,140) UBS Securities Australia Ltd Sell 2,304,978 (2,130) UBS Securities Australia Ltd Sell 2,304,978 (1,50) UBS Securities Australia Ltd Sell 2,304,978 (1,50) UBS Securities Australia Ltd Enter Into Derivative C 7,021 4,08 UBS Securities Australia Ltd Buy 2,304,978 (1,50) UBS Securities Australia Ltd Enter Into Derivative C 7,021		USS Securities Australia Ltd	Party Party	3.452	350	Ordings
UBS Securities Australia Ltd Buy 3.170 3.8 UBS Securities Australia Ltd Buy 4,690 500 UBS Securities Australia Ltd Buy 4,690 500 UBS Securities Australia Ltd Sell 1,0,672 (1,129) UBS Securities Australia Ltd Sell 23,700 (2,500) UBS Securities Australia Ltd Sell 23,700 (2,500) UBS Securities Australia Ltd Sell 23,049 (3,1149) UBS Securities Australia Ltd Sell 20,259 (2,1405) UBS Securities Australia Ltd Sell 2,304,978 (24,313) UBS Securities Australia Ltd Sell 2,304,978 (26,300) UBS Securities Australia Ltd Sell 2,304,978 (26,300) UBS Securities Australia Ltd Sell 2,504,978 (26,300) UBS Securities Australia Ltd Enter Into Derivative 7,021 (150) UBS Securities Australia Ltd Buy 2,304,978 (26,300) UBS Securities Australia Ltd Buy 2,504,978 (26,300		URS Securities Australia Ltd	Billy	2,551	500	Ordinary.
UBS Securities Australia Ltd Buy 4,690 500 UBS Securities Australia Ltd Buy 4,690 500 UBS Securities Australia Ltd Stock benowed NA 90,581 UBS Securities Australia Ltd Sell 1,1,291 UBS Securities Australia Ltd Sell 1,1,137 UBS Securities Australia Ltd Buy 197,137 20,632 UBS Securities Australia Ltd Buy 197,137 20,632 UBS Securities Australia Ltd Buy 197,137 20,632 UBS Securities Australia Ltd Sell 2,823 300 UBS Securities Australia Ltd Sell 2,304,978 (51,406) UBS Securities Australia Ltd Sell 2,522,189 (267,300) UBS Securities Australia Ltd Sell 2,522,189 (267,300) UBS Securities Australia Ltd Buy 1,722,350 (150) UBS Global Asset Management Buy 1,722,350 (1,500) UBS Global Asset Management Buy 1,22,361 (1,624) UBS Global Asset Management	Mav-14	UBS Securities Australia Ltd	Buv	3.170	338	Ordinary
UBS Securities Australia Ltd Buy 38 4 UBS Securities Australia Ltd Stock borrowed NA 90,951 UBS Securities Australia Ltd Sell 10,672 (1,129) UBS Securities Australia Ltd Sell 23,700 (2,500) UBS Securities Australia Ltd Sell 20,566 (21,149) UBS Securities Australia Ltd Sell 20,2,696 (21,466) UBS Securities Australia Ltd Sell 20,304,978 (24,34) UBS Securities Australia Ltd Sell 2,304,978 (24,34) UBS Securities Australia Ltd Sell 2,304,978 (26,578) UBS Securities Australia Ltd Sell 2,304,978 (26,578) UBS Securities Australia Ltd Enter into Derivative C 2,504,978 (26,578) UBS Securities Australia Ltd Buy 2,304,978 (150) UBS Securities Australia Ltd Enter into Derivative C 2,504,978 (26,578) UBS Securities Australia Ltd Buy 2,324,98 (26,578) UBS Global Asset Management Buy	May-14	UBS Securities Australia Ltd	Buv	4.690	200	Ordinary
UBS Securities Australia Ltd Stock borrowed NVA 90,951 UBS Securities Australia Ltd Sell 1,0572 (1,129) UBS Securities Australia Ltd Sell 23,700 (2,500) UBS Securities Australia Ltd Buy 19,137 20,832 UBS Securities Australia Ltd Sell 79,455 (21,406) UBS Securities Australia Ltd Sell 79,455 (8,14) UBS Securities Australia Ltd Sell 2,304,978 (24,14) UBS Securities Australia Ltd Sell 2,304,978 (26,780) UBS Securities Australia Ltd Sell 3,47,662 (36,578) UBS Securities Australia Ltd Enter Into Derivative 7,722 (150) UBS Securities Australia Ltd Enter Into Derivative 7,722 (150) UBS Securities Australia Ltd Enter Into Derivative 7,722 7,408 UBS Securities Australia Ltd Enter Into Derivative 7,722 1,704 UBS Securities Australia Ltd Buy 1,722,359 18,64 UBS Global Asset Management Buy<	May-14	UBS Securities Australia Etd	Boy	38	4	Ordinary
UBS Securities Australia Ltd Sell 1,129 UBS Securities Australia Ltd Sell 483.927 (1,129) UBS Securities Australia Ltd Sell 483.927 (5,140) UBS Securities Australia Ltd Buy 197,137 20,832 UBS Securities Australia Ltd Sell 2,925,596 (21,406) UBS Securities Australia Ltd Sell 2,304,978 (30,414) UBS Securities Australia Ltd Sell 3,747,622 (36,539) UBS Securities Australia Ltd Enter Into Derivative C 2,522,189 (267,300) UBS Securities Australia Ltd Enter Into Derivative C NA (150) UBS Global Asset Management Buy 1,722,359 (26,104) UBS Global Asset Management Buy 1,722,350 (16,24) UBS Global Asset Management Buy 1,722,381 (16,24) UBS Global Asset Management Buy 5,704 48,435 UBS Global Asset Management Buy 5,704 48,435 UBS Global Asset Management Buy 5,723,381 12,45,	Mav-14	UBS Securities Australia Ltd	Stock borrowed	AW	90 951	Ordinary
UBS Securities Australia Ltd Sell 22,700 (2,500) UBS Securities Australia Ltd Sell 4,250 (5,149) UBS Securities Australia Ltd Sell 7,945 (21,468) UBS Securities Australia Ltd Sell 7,945 (21,468) UBS Securities Australia Ltd Sell 7,945 (24,44) UBS Securities Australia Ltd Sell 2,304,978 (24,31) UBS Securities Australia Ltd Enter Into Derivative C 2,504,978 (26,300) UBS Securities Australia Ltd Enter Into Derivative C 2,504,978 (26,310) UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 237,964 25,01 UBS Global Asset Management Buy 53,01 5,671 UBS Global Asset Management Buy 53,01 </td <td>May-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>10 672</td> <td>(1 179)</td> <td>Ordinary</td>	May-14	UBS Securities Australia Ltd	Sell	10 672	(1 179)	Ordinary
UBS Securities Australia Ltd Sell 483,921 (51,149) UBS Securities Australia Ltd Buy 197,137 20,832 UBS Securities Australia Ltd Sell 79,455 (3,414) UBS Securities Australia Ltd Buy 230,457 (3,414) UBS Securities Australia Ltd Sell 2,304,978 (3,414) UBS Securities Australia Ltd Sell 3,47,662 (36,578) UBS Securities Australia Ltd Enter Into Derivative C 2,704,978 (150) UBS Securities Australia Ltd Enter Into Derivative C 7,702,1 1,702,1 UBS Securities Australia Ltd Enter Into Derivative C 7,702,1 1,702,1 UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 237,964 25,701 5,875 UBS Global Asset Management Buy 53,704 5,875 11,208 UBS Global Asset Management Buy 55,701 5,875 11,245,19 UBS Global Asset Management Buy 55,701 5,870 12,510	Vlay-14	UBS Securities Australia Ltd	lles.	23.700	(2.500)	Ordinary
UBS Securities Australia Ltd BLy 197,137 20,832 UBS Securities Australia Ltd Sell 20,2,696 (21,406) UBS Securities Australia Ltd BLy 2,304,978 300 UBS Securities Australia Ltd Sell 3,747,662 (243,913) UBS Securities Australia Ltd Sell 3,747,662 (265,300) UBS Securities Australia Ltd Enter Into Derivative C NA (150) UBS Securities Australia Ltd Enter Into Derivative C NA (150) UBS Global Asset Management Buy 2,522,189 (267,300) UBS Global Asset Management Buy 1,722,356 15,104 UBS Global Asset Management Buy 1,722,356 15,104 UBS Global Asset Management Buy 5,5701 5,653 UBS Global Asset Management Buy 5,5701 5,653 UBS Global Asset Management Buy 5,5701 5,601 UBS Global Asset Management Buy 5,5701 5,601 UBS Global Asset Management Buy 1,2,314 1,4,5	ı	UBS Securities Australia Ltd	Sell	483,921	(51.149)	Ordinary
UBS Securities Australia Ltd Sell 79,455 (21,406) UBS Securities Australia Ltd Sell 79,455 (8,414) UBS Securities Australia Ltd Sell 2,304,978 (243,913) UBS Securities Australia Ltd Sell 3,747,662 (36,578) UBS Securities Australia Ltd Enter Into Derivative C 2,522,169 (267,300) UBS Securities Australia Ltd Enter Into Derivative C 3,747,662 (36,578) UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 53,701 5,875 UBS Global Asset Management Buy 5,634 <td< td=""><td>1</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>197,137</td><td>20,832</td><td>Ordinary</td></td<>	1	UBS Securities Australia Ltd	Buy	197,137	20,832	Ordinary
UBS Securities Australia Ltd Sell 79,455 (8,414) UBS Securities Australia Ltd Buy 2,304,978 300 UBS Securities Australia Ltd Sell 3,747,662 (396,578) UBS Securities Australia Ltd Sell 3,747,662 (396,578) UBS Securities Australia Ltd Enter Into Derivative C 2,522,189 (267,300) UBS Securities Australia Ltd Enter Into Derivative C NAA (150) UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 1,722,359 181,664 UBS Global Asset Management Buy 53,701 5,875 UBS Global Asset Management Buy 53,1420 12,519 UBS Global Asset Management Buy 53,1420 12,51	ı	UBS Securities Australia Ltd	Sell	202,696	(21,406)	Ordinary
URB Securities Australa Ltd Buy 2.823 300 UBS Securities Australa Ltd Sell 2,304,978 (243,913) UBS Securities Australa Ltd Sell 3,747,662 (396,578) UBS Securities Australa Ltd Enter Into Derivative C 2,522,169 (267,300) UBS Securities Australa Ltd Enter Into Derivative C NAA (150) UBS Global Asset Management Buy 170,221 7,408 UBS Global Asset Management Buy 1,722,359 181,668 UBS Global Asset Management Buy 1,722,359 181,683 UBS Global Asset Management Buy 5,5701 5,683 UBS Global Asset Management Buy 5,5701 5,683 UBS Global Asset Management Buy 5,5701 5,873 UBS Global Asset Management Buy 1,180,566 112,519 UBS Global Asset Management Buy 1,180,566 112,519 UBS Global Asset Management Buy 1,180,566 112,519 UBS Global Asset Management Buy 1,180,566 <td< td=""><td>l</td><td>UBS Securities Australia Ltd</td><td>Seli</td><td>79,455</td><td>(8,414)</td><td>Ordinary</td></td<>	l	UBS Securities Australia Ltd	Seli	79,455	(8,414)	Ordinary
UBS Securities Australia Ltd Sell 2,304,978 (243,913) UBS Securities Australia Ltd Sell 3,747,662 (396,578) UBS Securities Australia Ltd Enter Into Derivative C 1,522,189 (150) UBS Securities Australia Ltd Enter Into Derivative C NAA (150) UBS Global Asset Naragement Buy 70,221 7,408 UBS Global Asset Naragement Buy 237,964 25,104 UBS Global Asset Management Buy 1,722,359 181,664 UBS Global Asset Management Buy 5,875 18,654 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 5,123 48,437 UBS Global Asset Management Buy 5,12,381 12,908 UBS Global Asset Management Buy 5,12,381 48,437 UBS Global Asset Management Buy 1,180,566 12,550 UBS Global Asset Management Buy 1,180,566 1,250 UBS Global Asset Management Buy 1,180,566 1,	Ŀ	UBS Securities Australia Ltd	Buy	2,823	300	Ordinary
USS Securities Australia Ltd Sell 3.747,662 (996,578) USS Securities Australia Ltd Sell 2,522,189 (267,300) UBS Securities Australia Ltd Enter Into Derivative C NAA (150) UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 1,722,359 181,664 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 53,701 5,875 UBS Global Asset Management Buy 53,1420 56,051 UBS Global Asset Management Buy 1,180,566 124,519 UBS Global Asset Management Buy 1,263,142 1,250	ı	UBS Securities Australia Ltd	Sell	2,304,978	(243.913)	Ordinary
UBS Securities Australia Ltd Seli 2,522,189 (267,300) UBS Securities Australia Ltd Enter Into Derivative (150) INA (150) UBS Global Asset Management (150) Buy 27,021 7,408 UBS Global Asset Management (150) Buy 1,702,359 18,164 UBS Global Asset Management (150) Buy 1,722,359 18,164 UBS Global Asset Management (150) Buy 5,570 5,673 UBS Global Asset Management (150) Buy 1,180,566 124,519 UBS Global Asset Management (150) Buy 1,180,566 124,519 UBS Global Asset Management (150) Buy 1,180,566 124,519 UBS Global Asset Management (150) Buy 1,23,14 2,510 UBS Global Asset Management (150) Bu	May-14	UBS Securities Australia Ltd	Sell	3,747,662	(336,578)	Ordinary
UBS Securities Australia Ltd Enter Into Derivative NAA (150) UBS Global Asset Management Buy 70,21 7,488 UBS Global Asset Management Buy 1,722,359 15,104 UBS Global Asset Management Buy 1,722,359 181,664 UBS Global Asset Management Buy 58,4356 25,104 UBS Global Asset Management Buy 58,4356 15,634 UBS Global Asset Management Buy 58,701 5,875 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 53,701 5,875 UBS Global Asset Management Buy 53,701 5,875 UBS Global Asset Management Buy 51,23,81 12,598 UBS Global Asset Management Buy 1,180,566 124,519 UBS Global Asset Management Buy 2,13,14 1,590 UBS Global Asset Management Buy 2,501 1,590 UBS Global Asset Management Buy 2,501 1,590 UBS Glob	May-14	UBS Securities Australia Ltd	ı	2,522,189	(267,300)	Ordinary
UBS Securities Australia Ltd Enter Into Derivative (NVA (150) UBS Global Asset Management Buy 70,221 7,408 UBS Global Asset Management Buy 237,964 25,108 UBS Global Asset Management Buy 1,722,339 181,664 UBS Global Asset Management Buy 53,01 5,875 UBS Global Asset Management Buy 53,01 5,875 UBS Global Asset Management Buy 53,01 5,875 UBS Global Asset Management Buy 53,701 5,875 UBS Global Asset Management Buy 53,701 5,875 UBS Global Asset Management Buy 1180,566 124,519 UBS Global Asset Management Buy 1,80,566 124,519 UBS Global Asset Management Buy 1,50,44 1,550 UBS Global Asset Management Buy 2,50,10 1,590 UBS Global Asset Management Buy 2,50,10 1,590 UBS Global Asset Management Buy 2,50,10 1,590 UBS Globa	May-14	UBS Securities Australia Ltd		AN	(150)	Options
UBS Global Asset Nanagement Buy 70.221 7.408 UBS Global Asset Nanagement Buy 237.964 25.104 UBS Global Asset Management Buy 1.722.359 181.664 UBS Global Asset Management Buy 564.256 61.634 UBS Global Asset Management Buy 55.701 5.875 UBS Global Asset Management Buy 1.23.381 12.908 UBS Global Asset Management Buy 1.80.566 124.519 UBS Global Asset Management Buy 1.80.566 124.519 UBS Global Asset Management Buy 1.50.91 1.500 UBS Global Asset Management Buy 2.63.14 2.580 UBS Global Asset Management Buy 2.63.14 2.800 UBS Global Asset Management Buy 2.63.14 2.910 UBS Global Asset Ma	May-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	(150)	Options
UBS Global Asset Management Buy 237,964 25,104 UBS Global Asset Management Buy 1,722,359 181,664 UBS Global Asset Management Buy 584,256 61,634 UBS Global Asset Management Buy 55,701 5,633 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 122,381 12,381 UBS Global Asset Management Buy 459,232 48,437 UBS Global Asset Management Buy 1,16,384 12,519 UBS Global Asset Management Buy 1,16,314 12,590 UBS Global Asset Management Buy 1,16,314 1,26,519 UBS Global Asset Management Buy 2,16,314 1,590 UBS Global Asset Management Buy 2,60,33 2,912 UBS Global Asset Management Buy 2,60,183 2,912 UBS Global Asset Management Buy 2,60,183 6,349 UBS Global Asset Management Buy 2,60,183 6,349 UBS Global	May-14	UBS Global Asset Management	Buy	70,221	7,408	Ordinary
UBS Global Asset Management Buy 1,722,359 181,664 UBS Global Asset Management Buy 5,633 61,634 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 53,420 5,875 UBS Global Asset Management Buy 51,420 56,031 UBS Global Asset Management Buy 7,180,566 124,519 UBS Global Asset Management Buy 7,180,566 124,519 UBS Global Asset Management Buy 7,180,566 124,519 UBS Global Asset Management Buy 7,603 2,912 UBS Global Asset Management Buy 2,603 2,912 UBS Global Asset Management Buy 2,603 2,912 UBS Global Asset Management Buy 7,603 2,912 UBS Global Asset Management Buy 7,603 2,912 UBS Global Asset Management Buy 7,603 6,183 UBS Global Asset Management<	May-14	UBS Global Asset Management	Buy	237,964	25,104	Ordinary
UBS Global Asset Management Buy 584,236 61,634 UBS Global Asset Management Buy 53,691 5,663 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 55,701 5,875 UBS Global Asset Management Buy 122,381 12,908 UBS Global Asset Management Buy 1,180,566 124,519 UBS Global Asset Management Buy 1,180,566 124,519 UBS Global Asset Management Buy 1,503,4 1,580 UBS Global Asset Management Buy 1,503,4 1,580 UBS Global Asset Management Buy 2,6314 1,580 UBS Global Asset Management Buy 1,503,4 1,580 UBS Global Asset Management Buy 2,6314 1,590 UBS Global Asset Management Buy 2,6314 1,590 UBS Global Asset Management Buy 2,613 2,912 UBS Global Asset Management Buy 2,601 3,912	May-14	UBS Global Asset Management	Buy	1,722,359	181,664	Ordinary
UBS Global Asset Management Buy 55.63 UBS Global Asset Management Buy 53.701 UBS Global Asset Management Buy 122.381 UBS Global Asset Management Buy 531.420 56.051 UBS Global Asset Management Buy 531.420 56.051 UBS Global Asset Management Buy 11.80,566 124.519 UBS Global Asset Management Buy 216.314 22.800 UBS Global Asset Management Buy 216.314 22.800 UBS Global Asset Management Buy 27.603 2.912	May-14	UBS Global Asset Management	Buy	584,236	61,634	Ordinary
UBS clobal Asset Management Buy 55,701 5,875 UBS clobal Asset Management Buy 122,381 12,308 UBS clobal Asset Management Buy 531,420 56,051 UBS clobal Asset Management Buy 459,232 48,437 UBS clobal Asset Management Buy 1,80,566 124,519 UBS clobal Asset Management Buy 2,6314 1,590 UBS clobal Asset Management Buy 2,603 2,912	May-14	UBS Global Asset Management	Buy	169'83	5,663	Ordinary
UBS Gobal Asset Nanagement Buy 123.381 12,908 UBS Gobal Asset Management Buy 531,420 56,051 UBS Gobal Asset Management Buy 459,232 48,437 UBS Gobal Asset Management Buy 1180,566 124,519 UBS Gobal Asset Management Buy 2,6314 1,509 UBS Gobal Asset Management Buy 2,6314 1,500 UBS Gobal Asset Management Buy 2,76314 1,500 UBS Gobal Asset Management Buy 2,912 6,183 UBS Gobal Asset Management Buy 2,912 6,183 UBS Gobal Asset Management Buy 2,912 1,500 UBS Gobal Asset Management Buy 6,0183 6,349 UBS Gobal Asset Management Buy 7,603 2,912	√ay-14	UBS Global Asset Management	Buy	55,701	5,875	Ordinary
UBS Global Asset Management Buy 55,314,200 56,051 UBS Global Asset Management Buy 1,180,566 124,519 UBS Global Asset Management Buy 1,180,566 124,519 UBS Global Asset Management Buy 2,63,14 2,820 UBS Global Asset Management Buy 1,5094 1,509 UBS Global Asset Management Buy 2,7,603 2,912 UBS Global Asset Management Buy 2,601 2,912 UBS Global Asset Management Buy 6,349 6,349 UBS Global Asset Management Buy 2,618 6,183 UBS Global Asset Management Buy 1,509 1,912	May-14	UBS Global Asset Management	Buy	122,381	12,908	Ordinary
UBS Global Asset Management Buy 459,232 48,437 UBS Global Asset Management Buy 1,180,566 124,519 UBS Global Asset Management Buy 216,314 22,820 UBS Global Asset Management Buy 15,094 1,590 UBS Global Asset Management Buy 27,603 2,912 UBS Global Asset Management Buy 2,613 2,912 UBS Global Asset Management Buy 7,601 1,590 UBS Global Asset Management Buy 7,601 1,590 UBS Global Asset Management Buy 7,701 1,701	May-14	UBS Global Asset Management	Buy	531,420	56,051	Ordinary
UBS Global Asset Management Buy 1,180,566 124,519 UBS Global Asset Management Buy 216,314 22,820 UBS Global Asset Management Buy 15,034 1,594 1,590 UBS Global Asset Management Buy 27,603 2,912 UBS Global Asset Management Buy 2,603 2,912 UBS Global Asset Management Buy 2,601 2,912 UBS Global Asset Management Buy 2,601 3,912 UBS Global Asset Management Buy 1,710 1,710	May-14	UBS Global Asset Management	Buy	459,232	48,437	Ordinary
UBS Global Asset Management Buy 216,314 22,820 UBS Global Asset Management Buy 15,094 1,590 UBS Global Asset Management Buy 27,603 2,912 UBS Global Asset Management Buy 60,183 6,349 IRS Washth Management Australia Inf Transferen NA 1 710	May-14	UBS Global Asset Management	Buy	1,180,566	124,519	Ordinary
UBS Global Asset Management Buy 1,590 UBS Global Asset Management Buy 27,603 2,912 UBS Global Asset Management Buy 60,183 6,6183 IRS Wealth Management Australia Inf Transferen NA 1,10	May-14	UBS Global Asset Management	Buy	216,314	22,820	Ordinary
UBS Global Asset Management Buy 27,603 2,912	Jay-14	UBS Global Asset Management	Buy	15,094	1,590	Ordinary
UBS Global Asset Management Buy 60,183 6,349 6,349 185 Wealth Management Australia Ind Transfer-In NA 1 219	May-14	UBS Global Asset Management	Buy	27,603	2,912	Ordinary
ILIBS Wealth Management Australia 10 Transfer-In 10 10 10 10 10 10 10 1	┪	UBS Global Asset Management	Buy	60,183	6,349	Ordinary
COS Wednit Management Australia Eta		CONTRACTOR AND CONTRACTOR OF THE PARTY OF TH				

26-May-14 26-May-14 26-May-14 26-May-14 26-May-14 26-May-14	UBS Securities Australia Ltd	Buy	313,678	33,375	Ordinary
26-May-14 26-May-14 26-May-14 26-May-14 26-May-14		And	0/5'0/		
26-May-14 26-May-14 26-May-14 26-May-14	Obsideration Authority and	1.5	100.00	010,0	Cidinaly
3y-14 3y-14	UBS Securities Australia Ltd	Seil	/66'08	(3,333)	Ordinary
3y-14 3y-14	UBS Securities Australia Ltd		83,405	(8,900)	Ordinary
26-May-14	UBS Securities Australia Ltd	Enter Into Derivative (ΝΑ	(150)	Options
1 T	UBS Securities Australia Ltd	Enter Into Derivative (NA	(150)	Options
	UBS Global Asset Management	Buy	178,009	18,903	Ordinary
26-May-14	UBS Global Asset Management	Buy	603,177	64,052	Ordinary
6-May-14	UBS Global Asset Management	Buy	1,480,870	157,255	Ordinary
6-May-14	UBS Global Asset Management	Buy	136,075	14,447	Ordinary
26-May-14	UBS Global Asset Management	Buy	141,170	14,988	Ordinary
26-May-14	UBS Global Asset Management	Buy	310,182	32,932	Ordinary
26-May-14	UBS Global Asset Management	Buy	1,347,012	143.012	Ordinary
av-14	UBS Global Asset Management	Buv	1,164,041	123.586	Ordinary
14-ye	I IRC Global Asset Management	Riny	2 007 405	217 703	Ordinan
26 May 14	IDC Clobal Acros Management	200	204,202	20111	Oldinary
Se han 14	UBC Clobal Acces Management	(ng	715 756 1	70,000	Oldinary
97-14	Upo Clobal Asset Ivial agentent	šina d	4,505,715	405,507	Orginary
O-Ividy-14	OBS GIODAI Asset Management	an)	197'05	4,059	Ordinary
6 hay 14	UBC Clobal Asset Management	, and	202,203	000 31	Oldinary
,	U.D. Maralet Administration Appendix 1 and	Ana	CCC,2C1	10,200	Oldinary
26-May-14	18C Countries Australia 14d	B 30	10 217	(1,304)	Countries
11/	DO Countition Australia (14)	, in a	0.217	250	Ordinary
6-May 14	I BS Societies Australia Ltd	200	CF0.3	(979)	Ordinal y
6-May-14	I IRC Societies Australia Ltd	100	0,072	(300)	Ordinary
6-May-14	I BS Societies Australia (14)	P	171,000	(15 212)	Ordinary
1 1	I IBC Cocurition A certain 14d	100	500,241	(7) 7(7)	Ordinary
26-May-14	IBS Societies Australia 144	list of	100	(CE3)	Ordinary
6 May 14	UDC Committee Australia Ltd	11-00	100,5	(700)	Oldinary
- A	ODD SECURIES AUSTRAIN LIG	100	570'/	(110)	Ordinary
O-IVIdy- (4	Obsideration Australia Lid	a s	49,47	(5,253)	Ordinary
	UBS Securities Australia Ltd	Sell	184,108	(19,586)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	470,000	(20,000)	Ordinary
26-May-14	UBS Securities Australia Ltd	Sell	4,644	(494)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	6,279	(668)	Ordinary
	UBS Securities Australia Ltd	Sell	7,614	(810)	Ordinary
	UBS Securities Australia Ltd	Sell	6,392	(089)	Ordinary
	UBS Securities Australia Ltd	Sell	127,436	(13,557)	Ordinary
Г	UBS Securities Australia Ltd	Sell	47,000	(2,000)	Ordinary
ay-14	UBS Securities Australia Ltd	Sell	23,970	(2.550)	Ordinary
26-May-14	UBS Securities Australia Ltd	Şe‡	177	(82)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	6.503	(1.011)	Ordinary
26-May-14	UBS Securities Australia Ltd	Sell	28	(3)	Ordinary
6-Mav-14	UBS Securities Australia Ltd	lle?	9.203	(626)	Ordinary
av-14	IBS Securities Australia I td	le Z	10.669	(1 135)	Ordinary
6-May-14	IRS Securities Australia Ind	ll S	00000	(000 //)	Ordinary
11/1	DO Coountion Australia 144	ll Coll	410,00	(4,044)	Ordinary
Т	OBS SECULIES Australia Liu	liac -	75,910	(4,757)	Ordinary
T	UBS Securities Australia Ltd	Ne le le le le le le le le le le le le le	40,035	(4,259)	Ordinary
1	UBS Securities Australia Ltd	Sell	585'Q	(101)	Ordinary
ay-14	UBS Securities Australia Ltd	Sell	4,850	(516)	Ordinary
26-May-14	UBS Securities Australia Ltd	Sell	25,427	(2,705)	Ordinary
ay-14	UBS Securities Australia Ltd	Seli	15,895	(1,691)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sett	17,512	(1,863)	Ordinary
6-May-14	UBS Securities Australia Ltd	Sell	25,982	(2,764)	Ordinary
Г	UBS Securities Australia Ltd	Sell	27 429	(2.918)	Ordinary
Т	UBS Segurities Australia I td	la ₂	26.499	(2,819)	Ordinary
Т	IBS Securities Australia Ltd	Z S	0 174	(976)	Ordinan
т	IBS Securities Australia (14	3	200	(0)	Ordinary
T	IRS Securities Australia I to	Sell	15 952	(1,697)	Ordinary
ı	IDC Cocuration Australia 144	100	200'C	(003)	Codingo
1	DS Securities Australia Ltd	liac S	4,900	(376)	Ordinary
20-Ivlay- :4	ODS Securities Australia LIO	Sell Sell	080'1	(010)	Ordinary
- 1	UBS Securities Australia Ltd	Seit Seit	1/7'5	(348)	Ordinary
T	UBS Securities Australia Ltd	Sell	1,090	(116)	Ordinary
7	UBS Securities Australia Ltd	Sell	1,090	(116)	Ordinary
	JBS Securities Australia Ltd	Sell	1,090	(116)	Ordinary
	UBS Securities Australia Ltd	les	12.652	(1.346)	Ordinary
- in in in in in in in in in in in in in	and open the street will be street		200,21	(0.50)	o dallia

33 Ordinary 574 Ordinary	L	229 Ordinary	L						╛	580 Ordinary	550 Ordinary	Ordinary		7,140 Ordinary			127 Ordinary	\perp	l63 Ordinary	13 Ordinary			111 Ordinary	2.125 Ordinary	L	31 Ordinary	100 Ordinary			129 Ordinary	376 Ordinary	L					736 Ordinary	103 Ordinary		6,677 Ordinary			93 Ordinary		_		146 Ordinary	L		_		434 Ordinary	L			147 Ordinary	4,362 Ordinary	
	2,738						686		9,420 1,0					67,259 7,1						1,064	7,338			20,018						1,215		1.884	1			-				62,897 6,6				1,206			1.375		Ε.		m	4,088					41,090 4,3	
								9						9		3								2									1							9									3	2	5						4	
d Buy		Buy								Buy			Buy										Bis								B Buy					Buy		3 Buy			Buy					Bally						Buy					J Buy	
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IBS Societities Australia 11	COO Secontiles Australia Ed	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Liv	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Lto	UBS Securities Australia Ltd	UBS Securities Australia Lto	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltv	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	The state of the state of the state of
28-May-14 28-May-14	28-May-14	28-May-14	28-May-14	28-May-14		20-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	28-May-14	1	11		28-May-14	28-May-14	28-Mav-14	28-May-14	28-May-14	28-May-14	1		1 1	ı	28-May-14								

27-May-14 27-May-14 27-May-14	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy	559	550	Ordinary
1 1	UBS Securities Australia Ltd	Buv	200	550	Ordinary
1					
	LIBS Securities Australia I to	N a	15.048	1 780	Ordinan
1	UBS Securities Australia Ltd	Buy	3 315	350	Ordinary
27-Mav-14	UBS Securities Australia I td	وا	4.659	(267)	Ordinary
27-May-14	I BC Counties Australia 14d	Sell	350 571	(55.5)	Ordinary
27-May-14	LIRS Sequirities Australia 14d	Sell Sell	530 OAE	(16,272)	Ordinary
27-May-14	LIBC Sequences Australia Ltd	Coll	350,043		Ordinary
27-1434-14	I IBC Con rition Australia 144	Coll	2,510	Т	Ordings
27 May 14	ODS Securities Australia Ltd	25.1	2,372,500	(000,062)	Ordinary
- 1	UBS Securities Australia Ltd	yes:	900,040,360	1	Ordinary
- 1	UBS Securities Australia Ltd	Sell	900,040,350	ı	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	208	22	Ordinary
	UBS Securities Australia Ltd	Buy	218	23	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	87	ε	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	094'6	1,000	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	711	12	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	814	98	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	946	100	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	7,237	165	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	5,591	591	Ordinary
27-May-14	UBS Securities Australia Ltd	Buy	10,292	1.088	Ordinary
27-Mav-14	UBS Securities Australia Ltd	Buv	293	31	Ordinary
27-Mav-14	UBS Securities Australia Ltd	Sell	141	(15)	Ordinary
28-May-14	UBS Securities Australia Ltd	Stock borrowed	ΑN	225,704	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	926'65	6.332	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	317,807	33,630	Ordinary
28-May-14	UBS Securities Australia Ltd	Sell	71,579	(2,606)	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	407,935	43,213	Ordinary
28-May-14	UBS Securities Australia Ltd	Sell	146,986	(15,548)	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	212,571	22,500	Ordinary
28-May-14	UBS Securities Australia Ltd	Sell	94,102	(10,000)	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	221,900	23,451	Ordinary
28-May-14	UBS Securities Australia Ltd	Sell	3,914,610	(410,315)	Ordinary
28-May-14	UBS Securities Australia Ltd	Sell	441,315	(46,257)	Ordinary
28-May-14	UBS Securities Australia Ltd	Sell	62,051	(6,504)	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	3,784,680	400,000	Ordinary
28-May-14	UBS Securities Australia Ltd	Enter Into Derivative C	N/A	(228)	Options
28-May-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	(150)	Options
28-May-14	UBS Securities Australia Ltd	Enter Into Derivative (ΑN	(20)	Options
28-May-14	UBS Global Asset Management	Buy	1,993	212	Ordinary
28-May-14	UBS Global Asset Management	Bry	1,144,986	121.807	Ordinary
28-May-14	UBS Global Asset Management	Buv	1.144.986	121.807	Ordinary
28-May-14	UBS Global Asset Management	Buy	1,144,986	121.807	Ordinary
28-May-14	UBS Global Asset Management	Buy	1,144,986	121,807	Ordinary
28-May-14	UBS Global Asset Management	Buv	1.144.986	121.807	Ordinary
28-Mav-14	UBS Global Asset Management	Buy	1.144.986	121.807	Ordinary
28-May-14	UBS Wealth Management Australia Ltd	Buv	45.780	4.847	Ordinary
28-May-14	UBS Wealth Management Australia I to	وا	1 388	(147)	Ordinary
28-Mav-14	UBS Wealth Management Australia Ltd	las.	117.066	(12.395)	Ordinary
28-May-14	UBS Securities Australia I td	llay	1819 660	(192 920)	Ordinary
28-May-14	I IBS Securities Australia I to	N. S.	216,266	23 007	Ordinary
	I IBS Securities Australia 14d	Buy	000 027	20,02	Ordinan
	1 DC Cognition Australia 144	in d	000,000	100,000	Ordinary
20-Ividy-14	Upo de minima Australia Liu	puy Pur.	14,030	#0C,1	Collially
	UDS SECURIES AUSTRAIN LIU	ony.	39,702	4,230	Ordinary
20-May-14	Obs Securities Australia Ltd	puy D	20,00	9,900	Ordinary
20-Iviay-14	Obs. Securities Australia Lid	buy D	701.7	+77	Clumary
20-May-14	UBS Securities Australia Ltd	Puy Bury	786	901	Ordinary
28-May-14	IRS Societies Australia Ind	B _{1.8}	100 0	057	Ordinary
28-May-14	I IBS Societies Australia I to	8111	100,01	1 810	Ordinan
29-May 14	130 Con which Australia 144	p o	201,71	373	Ordinary
28-May-14	LIRS Securities Australia Ltd	Bity	5,254	529	Ordinary
28-May-14	UBS Securities Australia Ltd	Buv	22.400	2.383	Ordinary
28-Mav-14	UBS Securities Australia Ltd	Buv	36,850	3,916	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	7 2 9 2 7	311	Ordinary
28-May-14	UBS Securities Australia Ltd	Buv	056	101	Ordinary

<u>할 저 첫 1로 된 기 </u>	20-IVIdy-14	I IDC Consistion Assets I and			2	Crainary
Bis Securities Autoriale LIG BMY 2,000 2,121 Bis Securities Autoriale LIG BMY 2,000 2,121 Bis Securities Autoriale LIG BMY 1,000 2,123 Bis Securities Autoriale LIG BMY 1,239 1462 Bis Securities Autoriale LIG BMY 2,753 2,887 Bis Securities Autoriale LIG BMY 2,753 3,774 Bis Securities Autoriale LIG BMY 2,783 3,774 Bis Securities Autoriale LIG BMY 2,784 2,784 Bis Securities Autoriale LIG BMY 2,784 2,784 Bis Securities Autoriale LIG <t< td=""><td>-May-14</td><td>I IRS Securities Australia Ltd</td><td>e de la companya de l</td><td>123</td><td>2</td><td>Ovoling</td></t<>	-May-14	I IRS Securities Australia Ltd	e de la companya de l	123	2	Ovoling
Big Securities Australia Ltd Buy 2,0019 2,124	Mav-14	UBS Securities Australia I1d	Buy	20 001	2 121	Ordinacy
USS Securities Autorials Ltd Buy 1,322 1,16 1,000 1,17 1,10 1,1	-May-14	I IRS Societifies Australia 14d	Brox	20,022	21.7	or dingra
OBS Securines Autominate to Buy Buy 1,2,202 2,116 OBS Securines Autominated Buy 1,2,323 1,8,86 OBS Securines Autominated Buy 7,85 2,86 OBS Securines Autominated Buy 2,7,53 2,88 OBS Securines Autominated Buy 2,7,53 2,88 OBS Securines Autominated Buy 2,7,53 2,88 OBS Securines Autominated Buy 2,5,541 2,7,64 OBS Securines Autominated Buy 2,5,541 2,7,64 OBS Securines Autominated Buy 2,6,73 3,67 OBS Securines Autominated Buy 2,6,73 3,77 OBS Securines Autominated Buy 2,6,25 3,60 OBS Securines Autominated Buy 2,6,25 3,60 OBS Securines Autominated Buy 2,6,25 <td>May 14</td> <td>Upo Securities Australia Ltd</td> <td>Soc</td> <td>90,02</td> <td>47174</td> <td>Ordinary</td>	May 14	Upo Securities Australia Ltd	Soc	90,02	47174	Ordinary
Obs. Securities Australia Ltd BMy 19,323 1,10 Obs. Securities Australia Ltd BMy 1,232 1,42 Obs. Securities Australia Ltd BMy 7,52 2,88 Obs. Securities Australia Ltd BMy 7,73 2,88 Obs. Securities Australia Ltd BMy 2,73 2,88 Obs. Securities Australia Ltd BMy 2,73 2,83 Obs. Securities Australia Ltd BMy 2,832 2,83 Obs. Securities Australia Ltd BMy 2,533 1,373 Obs. Securities Australia Ltd BMy 2,543 2,501 Obs. Securities Australia Ltd BMy 2,543 3,70 Obs. Securities Australia Ltd <	- Ividy- I	OBS Securities Australia Lid	Ang	500,7	213	Ordinary
Description of the control o	May 14	I IDC Cognition Appendix Lea	600	13,904	2,110	Countary
UBS Securities Australia Ltd Buy 1,239 1,42 UBS Securities Australia Ltd Buy 7,75 1,42 UBS Securities Australia Ltd Buy 7,72 2,88 1,42 UBS Securities Australia Ltd Buy 2,82 2,88 1,23 UBS Securities Australia Ltd Buy 5,63 1,23 30 UBS Securities Australia Ltd Buy 5,63 1,23 30 UBS Securities Australia Ltd Buy 5,63 1,23 30 UBS Securities Australia Ltd Buy 5,63 1,33 30 UBS Securities Australia Ltd Buy 5,63 1,34 4 UBS Securities Australia Ltd Buy 2,65 3,70 4,80 UBS Securities Australia Ltd Buy 2,65 3,70	May 14	I IDC Commission Assemble to	euy De	1,323	000,1	Ordinary
Obs. Securities Australia Ltd Buy 7.53 2.88 DBS Securities Australia Ltd Buy 2.7,243 2.88 DBS Securities Australia Ltd Buy 2.7,243 2.88 BBS Securities Australia Ltd Buy 2.7,243 2.88 BBS Securities Australia Ltd Buy 2.6,23 2.0 BBS Securities Australia Ltd Buy 5.6,33 2.0 BBS Securities Australia Ltd Buy 5.2 2.0 BBS Securities Australia Ltd Buy 5.2 3.0 BBS Securities Australia Ltd Buy 2.6 4.0 BBS Securities Australia Ltd Buy 2.6	May	I IBC Convision Australia Ltd	, in a	4,203	457	Ordinary
Obs. Securities Australia Ltd Buy 27,253 2,887 DES Securities Australia Ltd Buy 27,253 2,887 DES Securities Australia Ltd Buy 2,827 2,887 DES Securities Australia Ltd Buy 5,637 661 DES Securities Australia Ltd Buy 5,637 661 DES Securities Australia Ltd Buy 7,803 827 DES Securities Australia Ltd Buy 7,803 827 DES Securities Australia Ltd Buy 7,803 827 DES Securities Australia Ltd Buy 7,803 1,373 DES Securities Australia Ltd Buy 7,829 1,874 DES Securities Australia Ltd Buy 3,756 4,01 DES Securities Australia Ltd Buy	May 14	I IBC Cocurition Australia 144	bug bug	26	247	Ordinacy
USS Scornites Australia Ltd Buy 27,234 2,284 LDSS Scornites Australia Ltd Buy 27,544 2,248 LDSS Scornites Australia Ltd Buy 1,161 1,07 LDSS Scornites Australia Ltd Buy 1,161 1,07 LDSS Scornites Australia Ltd Buy 1,161 1,07 LDSS Scornites Australia Ltd Buy 1,273 1,273 LDSS Scornites Australia Ltd Buy 1,273 1,273 LDSS Scornites Australia Ltd Buy 2,827 500 LDSS Scornites Australia Ltd Buy 1,275 1,273 1,274 LDSS Scornites Australia Ltd Buy 2,825 2,001 1,001 LDSS Scornites Australia Ltd Buy 3,786 3,001 1,001 LDSS Scornites Australia Ltd Buy 3,786 3,001 1,001 LDSS Scornites Australia Ltd Buy 3,786 3,001 1,001 LDSS Scornites Australia Ltd Buy 3,248 1,001 1,001 LDSS Scornites Australia Ltd	Man 4	And Securities Advisagle Lid	ng	9/ 22	0 3	Ordinary
UBS Securities Australia Ltd Bly 5,274 2,748 UBS Securities Australia Ltd Bly 5,521 2,748 UBS Securities Australia Ltd Bly 5,523 661 UBS Securities Australia Ltd Bly 5,623 661 UBS Securities Australia Ltd Bly 7,673 1,373 UBS Securities Australia Ltd Bly 7,673 1,373 UBS Securities Australia Ltd Bly 7,673 1,373 UBS Securities Australia Ltd Bly 7,673 1,374 UBS Securities Australia Ltd Bly 7,673 1,474 UBS Securities Australia Ltd Bly 7,673 1,673 UBS Securities Australia Ltd Bly 3,776 4,00 UBS Securities Australia Ltd Bly <td>s-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>27,253</td> <td>2,887</td> <td>Ordinary</td>	s-May-14	UBS Securities Australia Ltd	Buy	27,253	2,887	Ordinary
UBS Securities Australia Ltd Bluy C/5-941 2/5-84 UBS Securities Australia Ltd Bluy 7/5-97 1.161 1.23 UBS Securities Australia Ltd Bluy 7,807 827 601 UBS Securities Australia Ltd Bluy 2,837 601 1.23 1.23 UBS Securities Australia Ltd Bluy 2,837 627 1.23 1.23 UBS Securities Australia Ltd Bluy 2,837 1.23 1.23 1.23 UBS Securities Australia Ltd Bluy 1,2551 1.434 1.23	-May-14	UBS Securities Australia Ltd	Buy	7/4	52	Ordinary
UBS Securities Auxitaria Ltd Bluy 514 66 UBS Securities Auxitaria Ltd Bluy 1,614 66 UBS Securities Auxitaria Ltd Bluy 2,632 300 UBS Securities Auxitaria Ltd Bluy 5,737 1,333 UBS Securities Auxitaria Ltd Bluy 1,2475 1,434 UBS Securities Auxitaria Ltd Bluy 2,525 2,601 UBS Securities Auxitaria Ltd Bluy 3,784 4,00 UBS Securities Auxitaria Ltd Bluy 3,786 4,00 UBS Securities Auxitaria Ltd Bluy 2,408 1,00 UBS Securities Auxitaria Ltd Bluy 2,40 1,00 UBS Securities Auxitaria Ltd B	-May-14	UBS Securities Australia Ltd	Buy	25,941	2,748	Ordinary
UBS Securities Australia Ltd BMy 1,161 113 UBS Securities Australia Ltd BMy 2,837 807 UBS Securities Australia Ltd BMy 2,837 807 UBS Securities Australia Ltd BMy 7,807 807 UBS Securities Australia Ltd BMy 1,297 1,373 UBS Securities Australia Ltd BMy 1,520 807 UBS Securities Australia Ltd BMy 3,814 4,00 UBS Securities Australia Ltd BMy 3,784 4,00 UBS Securities Australia Ltd BMy 4,00 4,00 UBS Securities Australia Ltd BMy 4,00 4,00 UBS Securities Australia Ltd BMy 4,00 4,00 UBS Securities Australia Ltd BMy 3,240 1,73 1,14 UBS Securities Australia Ltd <	s-May-14	UBS Securities Australia Ltd	Buy	614	92	Ordinary
UBS Securities Australia Ltd BMy 5,632 300 UBS Securities Australia Ltd Buy 5,632 300 UBS Securities Australia Ltd Buy 5,673 1,334 UBS Securities Australia Ltd Buy 1,297 1,434 UBS Securities Australia Ltd Buy 1,520 1,434 UBS Securities Australia Ltd Buy 2,525 2,601 UBS Securities Australia Ltd Buy 3,756 4,00 UBS Securities Australia Ltd Buy 2,635 1,14 UBS Securities Australia Ltd Buy 2,636 2,531 UBS Securities Australia Ltd Buy 2,636 2,531 UBS Securities Australia Ltd Buy	S-May-14	UBS Securities Australia Ltd	Buy	1,161	123	Ordinary
UBS Securities Australia Ltd Buy \$,673 601 UBS Securities Australia Ltd Buy 7,807 1,873 UBS Securities Australia Ltd Buy 7,807 1,873 UBS Securities Australia Ltd Buy 1,874 1,874 UBS Securities Australia Ltd Buy 3,814 4,02 UBS Securities Australia Ltd Buy 3,814 4,02 UBS Securities Australia Ltd Buy 3,814 4,00 UBS Securities Australia Ltd Buy 3,82 87 UBS Securities Australia Ltd Buy 2,08 87 UBS Securities Australia Ltd Buy 3,07 4,02 UBS Securities Australia Ltd Buy 3,09 1,03 UBS Securities Australia Ltd Buy 3,00 1,03 UBS Securities Australia Ltd Buy <t< td=""><td>3-May-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>2,832</td><td>300</td><td>Ordinary</td></t<>	3-May-14	UBS Securities Australia Ltd	Buy	2,832	300	Ordinary
UBS Securities Australia Ltd Buy 7,807 827 UBS Securities Australia Ltd Buy 12,975 1,373 UBS Securities Australia Ltd Buy 1,2,975 1,373 UBS Securities Australia Ltd Buy 2,4,553 2,601 UBS Securities Australia Ltd Buy 3,814 4,00 UBS Securities Australia Ltd Buy 3,276 4,00 UBS Securities Australia Ltd Buy 2,601 1,00 UBS Securities Australia Ltd Buy </td <td>3-May-14</td> <td>IUBS Securities Australia Ltd</td> <td>Buy</td> <td>5,673</td> <td>601</td> <td>Ordinary</td>	3-May-14	IUBS Securities Australia Ltd	Buy	5,673	601	Ordinary
UBS Securities Australia Ltd Buy 13.73 UBS Securities Australia Ltd Buy 13.73 1.43 UBS Securities Australia Ltd Buy 52,657 6 UBS Securities Australia Ltd Buy 24,553 2,601 UBS Securities Australia Ltd Buy 32,828 3,479 UBS Securities Australia Ltd Buy 3,776 400 UBS Securities Australia Ltd Buy 3,776 400 UBS Securities Australia Ltd Buy 3,776 400 UBS Securities Australia Ltd Buy 1,520 1,613 UBS Securities Australia Ltd Buy 3,776 48 UBS Securities Australia Ltd Buy 1,535 1,613 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 4,536 1,613 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 2,532 2,61 UBS Securities Australia Ltd Buy 1,731	3-May-14	UBS Securities Australia Ltd	Buy	7,807	827	Ordinary
UBS Securities Australia Ltd Buy 13,551 1,434 UBS Securities Australia Ltd Buy 13,551 1,434 UBS Securities Australia Ltd Buy 24,553 2,601 UBS Securities Australia Ltd Buy 32,842 3479 UBS Securities Australia Ltd Buy 3,814 404 UBS Securities Australia Ltd Buy 3,726 409 UBS Securities Australia Ltd Buy 3,727 33 UBS Securities Australia Ltd Buy 2,081 1,737 184 UBS Securities Australia Ltd Buy 2,081 2,551 167 UBS Securities Australia Ltd Buy 2,408 1,737 184 UBS Securities Australia Ltd Buy 2,408 1,703 48 UBS Securities Australia Ltd Buy 2,409 1,703 48 UBS Securities Australia Ltd Buy 2,409 1,703 48 UBS Securities Australia Ltd Buy 2,409 1,703 48 48 UBS Securities A	3-May-14	UBS Securities Australia Ltd	Buy	12,975	1,373	Ordinary
UBS Securities Australia Ltd Buy 57 6 UBS Securities Australia Ltd Buy 54553 2 601 UBS Securities Australia Ltd Buy 24,553 2 601 UBS Securities Australia Ltd Buy 3,776 400 UBS Securities Australia Ltd Buy 4,624 4,03 UBS Securities Australia Ltd Buy 4,63 4,63 UBS Securities Australia Ltd Buy 4,73 4,83 UBS Securities Australia Ltd Buy 4,73 4,83 UBS Securities Australia Ltd Buy 4,73 4,83 UBS Securities Australia Ltd Buy 1,737 1,83 UBS Securities Australia Ltd Buy 2,530 1,737 UBS Securities Australia Ltd Buy 2,	-May-14	IUBS Securities Australia Ltd	Buy	13,551	1,434	Ordinary
UBS Securities Australia Ltd Buy 24.553 2.61 UBS Securities Australia Ltd Buy 24.553 2.61 UBS Securities Australia Ltd Buy 32.842 3.479 UBS Securities Australia Ltd Buy 3.78 4.02 UBS Securities Australia Ltd Buy 3.77 4.02 UBS Securities Australia Ltd Buy 2.0,845 2.07 UBS Securities Australia Ltd Buy 2.0,841 4.00 4.00 UBS Securities Australia Ltd Buy 2.081 3.00 1.14 UBS Securities Australia Ltd Buy 2.031 3.23 UBS Securities Australia Ltd Buy 2.031 3.00 UBS Securities Australia Ltd Buy 2.021 3.00	3-May-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
UBS Securities Australia Ltd Buy 3,4553 2,601 UBS Securities Australia Ltd Buy 3,2842 3,499 UBS Securities Australia Ltd Buy 3,776 400 UBS Securities Australia Ltd Buy 3,776 400 UBS Securities Australia Ltd Buy 3,776 400 UBS Securities Australia Ltd Buy 2,0,845 2,077 UBS Securities Australia Ltd Buy 4,6,266 409 UBS Securities Australia Ltd Buy 4,6,261 109 UBS Securities Australia Ltd Buy 4,529 1,613 UBS Securities Australia Ltd Buy 7,701 48 UBS Securities Australia Ltd Buy 2,531 1,701 UBS Securities Australia Ltd Buy	8-May-14	UBS Securities Australia Ltd	Buy	57	9	Ordinary
UBS Securities Australia Ltd Buy 3.2842 3.479 UBS Securities Australia Ltd Buy 3.716 4.00 UBS Securities Australia Ltd Buy 3.776 4.00 UBS Securities Australia Ltd Buy 2.0,845 2.207 UBS Securities Australia Ltd Buy 2.0,845 2.207 UBS Securities Australia Ltd Buy 2.0,845 2.507 UBS Securities Australia Ltd Buy 2.0,845 2.507 UBS Securities Australia Ltd Buy 2.0,845 3.2 UBS Securities Australia Ltd Buy 2.0,845 3.2 UBS Securities Australia Ltd Buy 2.0,840 3.0 UBS Securities Australia Ltd	3-May-14	UBS Securities Australia Ltd	Buy	24,553	2,601	Ordinary
UBS Securities Australia Ltd Buy 3.814 402 UBS Securities Australia Ltd Buy 3.776 400 UBS Securities Australia Ltd Buy 3.776 400 UBS Securities Australia Ltd Buy 1.520 161 UBS Securities Australia Ltd Buy 2.0,885 2.207 UBS Securities Australia Ltd Buy 8,20,885 1.613 UBS Securities Australia Ltd Buy 8,20,885 1.613 UBS Securities Australia Ltd Buy 2,600 1.73 UBS Securities Australia Ltd Buy 4,636 2,501 UBS Securities Australia Ltd Buy 4,731 1.88 UBS Securities Australia Ltd Buy 2,637 2,54 UBS Securities Australia Ltd Buy 2,637 2,54 UBS Securities Australia Ltd Buy 2,637 2,54 UBS Securities Australia Ltd Buy 2,531 1,68 UBS Securities Australia Ltd Buy 2,531 1,632 UBS Securities Australia Ltd Buy	3-May-14	UBS Securities Australia Ltd	Buy	32,842	3,479	Ordinary
UBS Securities Australia Ltd Buy 3776 400 UBS Securities Australia Ltd Buy 3176 400 UBS Securities Australia Ltd Buy 1,520 161 UBS Securities Australia Ltd Buy 4,626 2,607 UBS Securities Australia Ltd Buy 4,626 2,607 UBS Securities Australia Ltd Buy 4,626 2,607 UBS Securities Australia Ltd Buy 1,613 1,613 UBS Securities Australia Ltd Buy 3,608 1,613 UBS Securities Australia Ltd Buy 4,701 4,88 UBS Securities Australia Ltd Buy 4,701 4,88 UBS Securities Australia Ltd Buy 4,701 4,88 UBS Securities Australia Ltd Buy 5,232 2,47 UBS Securities Australia Ltd Buy 5,232 2,51 UBS Securities Australia Ltd Buy 1,731 1,14 UBS Securities Australia Ltd Buy 2,332 2,44 UBS Securities Australia Ltd Buy	3-May-14	UBS Securities Australia Ltd	Buy	3,814	404	Ordinary
UBS Securities Australia Ltd Buy 1,520 161 UBS Securities Australia Ltd Buy 1,520 161 UBS Securities Australia Ltd Buy 2,0,845 2,207 161 UBS Securities Australia Ltd Buy 2,0,845 1,53 1,613 UBS Securities Australia Ltd Buy 1,737 184 UBS Securities Australia Ltd Buy 2,4,881 2,551 UBS Securities Australia Ltd Buy 2,4,881 2,551 UBS Securities Australia Ltd Buy 2,6,37 4,88 UBS Securities Australia Ltd Buy 4,701 4,88 UBS Securities Australia Ltd Buy 5,636 5,97 UBS Securities Australia Ltd Buy 5,637 2,24 UBS Securities Australia Ltd Buy 1,731 1,18 UBS Securities Australia Ltd Buy 1,531 1,418 UBS Securities Australia Ltd Buy 1,531 1,431 UBS Securities Australia Ltd Buy 1,532 2,44 UBS	8-May-14	UBS Securities Australia Ltd	Buy	3,776	400	Ordinary
UBS Securities Australia Ltd Buy 1520 UBS Securities Australia Ltd Buy 20,845 2,077 UBS Securities Australia Ltd Buy 4,625 4,90 UBS Securities Australia Ltd Buy 4,625 1,613 UBS Securities Australia Ltd Buy 1,024 1,1 UBS Securities Australia Ltd Buy 2,62,081 2,521 UBS Securities Australia Ltd Buy 1,076 48 UBS Securities Australia Ltd Buy 7,01 48B UBS Securities Australia Ltd Buy 7,07 48B UBS Securities Australia Ltd Buy 5,23 2,44 UBS Securities Australia Ltd Buy 5,23 2,44 UBS Securities Australia Ltd Buy 2,332 2,44 UBS Securities Australia Ltd Buy 2,532 2,44 UBS Securities Australia Ltd Buy 2,532 2,44 UBS Securities Australia Ltd Buy 1,273 1,48 UBS Securities Australia Ltd Buy 1,273	5-May-14	UBS Securities Australia Ltd	Buy	312	33	Ordinary
UBS Securities Australia Ltd Buy 20,845 2,207 UBS Securities Australia Ltd Buy 4,226 4,99 UBS Securities Australia Ltd Buy 8,269 8,76 UBS Securities Australia Ltd Buy 8,269 8,76 UBS Securities Australia Ltd Buy 2,608 2,551 UBS Securities Australia Ltd Buy 2,608 2,551 UBS Securities Australia Ltd Buy 4,701 4,98 UBS Securities Australia Ltd Buy 4,701 4,98 UBS Securities Australia Ltd Buy 4,701 4,98 UBS Securities Australia Ltd Buy 5,23 5,81 UBS Securities Australia Ltd Buy 5,536 2,07 UBS Securities Australia Ltd Buy 1,731 1,83 UBS Securities Australia Ltd Buy 2,531 2,637 UBS Securities Australia Ltd Buy 1,731 1,84 UBS Securities Australia Ltd Buy 1,731 1,731 UBS Securities Australia Ltd Buy <td>-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>1,520</td> <td>161</td> <td>Ordinary</td>	-May-14	UBS Securities Australia Ltd	Buy	1,520	161	Ordinary
UBS Securities Australia Ltd Buy 4,626 490 UBS Securities Australia Ltd Buy 1,535 1,613 UBS Securities Australia Ltd Buy 8,260 11 UBS Securities Australia Ltd Buy 2,600 17 UBS Securities Australia Ltd Buy 2,600 1,84 UBS Securities Australia Ltd Buy 2,600 1,137 UBS Securities Australia Ltd Buy 2,701 4,88 UBS Securities Australia Ltd Buy 4,701 4,89 UBS Securities Australia Ltd Buy 7,61 8,9 UBS Securities Australia Ltd Buy 5,536 5,97 UBS Securities Australia Ltd Buy 5,536 5,97 UBS Securities Australia Ltd Buy 2,731 1,88 UBS Securities Australia Ltd Buy 2,532 2,44 UBS Securities Australia Ltd Buy 2,532 1,323 UBS Securities Australia Ltd Buy 2,532 1,324 UBS Securities Australia Ltd Buy	-May-14	UBS Securities Australia Ltd	Buv	20.845	2.207	Ordinary
UBS Securities Australia Ltd Buy 15,235 1,613 UBS Securities Australia Ltd Buy 8,269 876 UBS Securities Australia Ltd Buy 104 114 UBS Securities Australia Ltd Buy 2,681 2,551 UBS Securities Australia Ltd Buy 4,701 488 UBS Securities Australia Ltd Buy 4,701 488 UBS Securities Australia Ltd Buy 4,701 488 UBS Securities Australia Ltd Buy 7,73 144 UBS Securities Australia Ltd Buy 5,231 2,40 UBS Securities Australia Ltd Buy 5,536 5,97 UBS Securities Australia Ltd Buy 1,076 1,32 UBS Securities Australia Ltd Buy 2,531 1,48 UBS Securities Australia Ltd Buy 1,731 1,18 UBS Securities Australia Ltd Buy 1,532 1,44 UBS Securities Australia Ltd Buy 1,731 1,18 UBS Securities Australia Ltd Buy <	5-May-14	UBS Securities Australia Ltd	Buy	4.626	767	Ordinary
UBS Securities Australia Ltd Buy 8,559 175 UBS Securities Australia Ltd Buy 302 32 UBS Securities Australia Ltd Buy 24,081 2,551 UBS Securities Australia Ltd Buy 4,301 488 UBS Securities Australia Ltd Buy 4,701 488 UBS Securities Australia Ltd Buy 4,701 488 UBS Securities Australia Ltd Buy 4,701 498 UBS Securities Australia Ltd Buy 5,636 5,97 UBS Securities Australia Ltd Buy 5,636 2,09 UBS Securities Australia Ltd Buy 1,000 1,000 UBS Securities Australia Ltd Buy 1,531 2,47 UBS Securities Australia Ltd Buy 2,537 2,47 UBS Securities Australia Ltd Buy 2,537 2,47 UBS Securities Australia Ltd Buy 1,731 1,83 UBS Securities Australia Ltd Buy 1,731 1,83 UBS Securities Australia Ltd Buy <	-Mav-14	ILIBS Securities Australia I td	Bix	15,235	1 613	Ordinary
UBS Securities Australia Ltd Buy 104 11 UBS Securities Australia Ltd Buy 302 32 UBS Securities Australia Ltd Buy 24,081 2,531 UBS Securities Australia Ltd Buy 453 48 UBS Securities Australia Ltd Buy 7,737 184 UBS Securities Australia Ltd Buy 7,701 488 UBS Securities Australia Ltd Buy 7,701 488 UBS Securities Australia Ltd Buy 5,736 5,700 UBS Securities Australia Ltd Buy 5,736 5,700 UBS Securities Australia Ltd Buy 5,732 2,44 UBS Securities Australia Ltd Buy 5,732 2,44 UBS Securities Australia Ltd Buy 2,732 2,44 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 2,722 1,44 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 2,	May-14	I IRS Securities Australia 14d	(5)	0.000	320	Ordings
UBS Securities Australia Ltd Buy 302 31 UBS Securities Australia Ltd Buy 26.081 2.551 UBS Securities Australia Ltd Buy 4,701 408 UBS Securities Australia Ltd Buy 4,701 408 UBS Securities Australia Ltd Buy 4,701 408 UBS Securities Australia Ltd Buy 9,195 9,4 UBS Securities Australia Ltd Buy 9,400 1,000 UBS Securities Australia Ltd Buy 5,536 597 UBS Securities Australia Ltd Buy 1,076 1,14 UBS Securities Australia Ltd Buy 1,332 2,47 UBS Securities Australia Ltd Buy 1,532 2,51 UBS Securities Australia Ltd Buy 1,731 1,18 UBS Securities Australia Ltd Buy <	-May-14	I IRC Securities Australia 1+d	, and	104	200	Ordinary
UBS Securities Australia Ltd Buy 24,081 2551 UBS Securities Australia Ltd Buy 24,081 2551 UBS Securities Australia Ltd Buy 4701 488 UBS Securities Australia Ltd Buy 4,701 498 UBS Securities Australia Ltd Buy 7,73 114 UBS Securities Australia Ltd Buy 9,40 1,000 UBS Securities Australia Ltd Buy 5,29 14 UBS Securities Australia Ltd Buy 5,636 1,14 UBS Securities Australia Ltd Buy 2,532 2,637 UBS Securities Australia Ltd Buy 2,532 1,738 UBS Securities Australia Ltd Buy 2,532 1,738 UBS Securities Australia Ltd Buy	May 14	Library Australia Library	Sod Pool	101	~ 6	Oldinary
UBS Securities Australia Ltd Buy 24,081 2,551 UBS Securities Australia Ltd Buy 1,737 48 UBS Securities Australia Ltd Buy 4,731 48 UBS Securities Australia Ltd Buy 7,6 8 UBS Securities Australia Ltd Buy 5,2 5,7 UBS Securities Australia Ltd Buy 5,2 5,6 UBS Securities Australia Ltd Buy 5,635 5,9 UBS Securities Australia Ltd Buy 2,635 2,0 UBS Securities Australia Ltd Buy 2,635 2,0 UBS Securities Australia Ltd Buy 1,731 1,8 UBS Securities Australia Ltd Buy 1,2,20 1,5,24 UBS Securities Australia Ltd Buy 1,2,20 <td>-Ividy-14</td> <td>OBS Securities Australia LIG</td> <td>gnà</td> <td>305</td> <td>37</td> <td>Ordinary</td>	-Ividy-14	OBS Securities Australia LIG	gnà	305	37	Ordinary
UBS Securities Australia Ltd Buy 4,701 184 UBS Securities Australia Ltd Buy 4,701 498 UBS Securities Australia Ltd Buy 4,701 498 UBS Securities Australia Ltd Buy 9,195 9,195 UBS Securities Australia Ltd Buy 9,195 9,100 UBS Securities Australia Ltd Buy 5,536 597 UBS Securities Australia Ltd Buy 1,076 1,14 UBS Securities Australia Ltd Buy 2,332 2,47 UBS Securities Australia Ltd Buy 1,731 1,83 UBS Securities Australia Ltd Buy 1,731 1,620 UBS Securities Australia Ltd Buy 1,731 1,620 UBS Securities Australia Ltd Buy	-Ividy-	Obsideration Australia Ltd	poo	24,081	7,551	Ordinary
UBS Securities Australia Ltd Buy 453 48 UBS Securities Australia Ltd Buy 4,701 48 UBS Securities Australia Ltd Buy 9,195 974 UBS Securities Australia Ltd Buy 9,440 1,000 UBS Securities Australia Ltd Buy 5,23 56 UBS Securities Australia Ltd Buy 5,536 56 UBS Securities Australia Ltd Buy 2,531 266 UBS Securities Australia Ltd Buy 2,532 2,636 UBS Securities Australia Ltd Buy 2,532 2,637 UBS Securities Australia Ltd Buy 2,532 2,79 UBS Securities Australia Ltd Buy 2,532 1,731 UBS Securities Australia Ltd Buy 2,128 2,44 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 4,290 4,54 UBS Securities Australia Ltd Buy 1,731 1,731 UBS Securities Australia Ltd Buy <td< td=""><td>-May-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1,73/</td><td>184</td><td>Ordinary</td></td<>	-May-14	UBS Securities Australia Ltd	Buy	1,73/	184	Ordinary
UBS Securities Australia Ltd Buy 4,701 498 UBS Securities Australia Ltd Buy 7,75 97.4 UBS Securities Australia Ltd Buy 9,195 97.4 UBS Securities Australia Ltd Buy 5,29 4,40 1,000 UBS Securities Australia Ltd Buy 5,536 5,97 UBS Securities Australia Ltd Buy 2,511 2,66 UBS Securities Australia Ltd Buy 2,637 2,73 UBS Securities Australia Ltd Buy 2,637 1,73 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 1,731 1,90 UBS Securities Australia Ltd Buy 1,731 1,50 UBS Securities Australia Ltd Buy 1,239 2,031 UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Buy	453	48	Ordinary
UBS Securities Australia Ltd Buy 76 8 UBS Securities Australia Ltd Buy 9,195 9,44 UBS Securities Australia Ltd Buy 5,236 5,69 UBS Securities Australia Ltd Buy 1,076 5,731 266 UBS Securities Australia Ltd Buy 5,631 2,67 2,67 UBS Securities Australia Ltd Buy 1,531 267 2,74 UBS Securities Australia Ltd Buy 2,332 2,47 2,637 2,74 UBS Securities Australia Ltd Buy 1,731 1,88 2,24 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 1,731 1,731 1,88 UBS Securities Australia Ltd Buy 1,731 1,731 1,731 1,731 1,731 1,731 1,731 1,731 1,731 1,731 1,731 1,731 1,731 1,732 1,731 1,731 <t< td=""><td>-May-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>4,701</td><td>498</td><td>Ordinary</td></t<>	-May-14	UBS Securities Australia Ltd	Buy	4,701	498	Ordinary
UBS Securities Australia Ltd Buy 9.195 974 UBS Securities Australia Ltd Buy 9,440 1,000 UBS Securities Australia Ltd Buy 529 56 UBS Securities Australia Ltd Buy 5,535 5,97 UBS Securities Australia Ltd Buy 2,511 2,66 UBS Securities Australia Ltd Buy 2,332 247 UBS Securities Australia Ltd Buy 2,332 247 UBS Securities Australia Ltd Buy 2,332 247 UBS Securities Australia Ltd Buy 1,731 183 UBS Securities Australia Ltd Buy 2,122 2,44 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 1,250 1,188 UBS Securities Australia Ltd Buy 1,239 2,64 UBS Securities Australia Ltd Buy 1,227 1,188 UBS Securities Australia Ltd Buy 1,239 2,61 UBS Securities Australia Ltd Buy <	-May-14	UBS Securities Australia Ltd	Buy	9/	8	Ordinary
UBS Securities Australia Ltd Buy 9,400 1,000 UBS Securities Australia Ltd Buy 5,29 56 UBS Securities Australia Ltd Buy 5,536 597 UBS Securities Australia Ltd Buy 2,331 2,27 UBS Securities Australia Ltd Buy 2,332 2,47 UBS Securities Australia Ltd Buy 2,332 2,47 UBS Securities Australia Ltd Buy 1,731 1,83 UBS Securities Australia Ltd Buy 1,731 1,50 UBS Securities Australia Ltd Buy 1,731 1,50 UBS Securities Australia Ltd Buy 1,20 1,53 UBS Securities Australia Ltd Buy 1,20 1,62 UBS Securities Australia Ltd Buy 1,42 1,5 UBS Securities Australia Ltd Buy <t< td=""><td>-May-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>9,195</td><td>974</td><td>Ordinary</td></t<>	-May-14	UBS Securities Australia Ltd	Buy	9,195	974	Ordinary
UBS Securities Australia Ltd Buy 529 56 UBS Securities Australia Ltd Buy 5.036 5.93 UBS Securities Australia Ltd Buy 5.631 2.66 UBS Securities Australia Ltd Buy 2.337 2.77 UBS Securities Australia Ltd Buy 2.637 2.79 UBS Securities Australia Ltd Buy 2.637 2.74 UBS Securities Australia Ltd Buy 1.731 1.83 UBS Securities Australia Ltd Buy 2.1.228 2.244 UBS Securities Australia Ltd Buy 1.732 1.188 UBS Securities Australia Ltd Buy 4.290 4.54 UBS Securities Australia Ltd Buy 4.290 4.54 UBS Securities Australia Ltd Buy 4.290 4.54 UBS Securities Australia Ltd Buy 1.5384 1.676 UBS Securities Australia Ltd Buy 2.2.538 2.88 UBS Securities Australia Ltd Buy 2.2.538 2.88 UBS Securities Australia Ltd Buy <td>-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>9.440</td> <td>1.000</td> <td>Ordinary</td>	-May-14	UBS Securities Australia Ltd	Buy	9.440	1.000	Ordinary
UBS Securities Australia Ltd Buy 1,076 1,076 1,14 UBS Securities Australia Ltd Buy 2,513 266 UBS Securities Australia Ltd Buy 2,513 2,637 UBS Securities Australia Ltd Buy 2,637 2,79 UBS Securities Australia Ltd Buy 1,502 1,323 UBS Securities Australia Ltd Buy 1,731 183 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 1,731 1,88 UBS Securities Australia Ltd Buy 1,418 1,50 UBS Securities Australia Ltd Buy 1,418 1,50 UBS Securities Australia Ltd Buy 1,534 1,53 UBS Securities Australia Ltd Buy 1,534 1,63 UBS Securities Australia Ltd Buy 1,23 2,031 UBS Securities Australia Ltd Buy 1,23 2,53 UBS Securities Australia Ltd Buy 1,427 151 UBS Securities Australia Ltd	-May-14	I IRS Securities Australia 11d	Bity	520	95	Ordinary
UBS Securities Australia Ltd Buy 5.636 5.94 UBS Securities Australia Ltd Buy 5.636 5.95 UBS Securities Australia Ltd Buy 2.511 2.60 UBS Securities Australia Ltd Buy 2.637 2.73 UBS Securities Australia Ltd Buy 2.637 2.73 UBS Securities Australia Ltd Buy 1,731 183 UBS Securities Australia Ltd Buy 2,1,228 2,244 UBS Securities Australia Ltd Buy 1,1,227 1,188 UBS Securities Australia Ltd Buy 4,290 4,24 UBS Securities Australia Ltd Buy 1,217 1,188 UBS Securities Australia Ltd Buy 1,233 1,50 UBS Securities Australia Ltd Buy 1,319 2,23 UBS Securities Australia Ltd Buy 1,427 1,51 UBS Securities Australia Ltd Buy 2,538 1,676 UBS Securities Australia Ltd Buy 2,538 2,385 UBS Securities Australia Ltd Buy<	Man 14	Library Actions Co.	Sing.	926	8	Galnary
UBS Securities Australia Ltd Buy 5,635 597 UBS Securities Australia Ltd Buy 2,611 2,666 UBS Securities Australia Ltd Buy 18,966 2,009 UBS Securities Australia Ltd Buy 2,633 247 UBS Securities Australia Ltd Buy 1,502 1,323 UBS Securities Australia Ltd Buy 2,234 1,88 UBS Securities Australia Ltd Buy 1,731 183 UBS Securities Australia Ltd Buy 1,1,227 1,188 UBS Securities Australia Ltd Buy 4,290 4,54 UBS Securities Australia Ltd Buy 1,132 1,62 UBS Securities Australia Ltd Buy 1,239 1,67 UBS Securities Australia Ltd Buy 1,230 1,67 UBS Securities Australia Ltd Buy	-IMIGN-14	ODS Securities Australia Ltd	Buy	9/0/1	1:4	Ordinary
UBS Securities Australia Ltd Buy 2,311 266 UBS Securities Australia Ltd Buy 2,332 247 UBS Securities Australia Ltd Buy 2,637 273 UBS Securities Australia Ltd Buy 1,731 183 UBS Securities Australia Ltd Buy 1,731 1,84 UBS Securities Australia Ltd Buy 1,731 1,84 UBS Securities Australia Ltd Buy 1,731 1,84 UBS Securities Australia Ltd Buy 1,731 1,80 UBS Securities Australia Ltd Buy 1,731 1,50 UBS Securities Australia Ltd Buy 1,200 454 UBS Securities Australia Ltd Buy 1,673 2,031 UBS Securities Australia Ltd Buy 1,427 1,57 UBS Securities Australia Ltd Buy	to IMIAy	UBS Securities Australia Ltd	Buy	gra/c	765	Ordinary
UBS Securities Australia Ltd Buy 18,955 2,009 UBS Securities Australia Ltd Buy 2,837 2,73 UBS Securities Australia Ltd Buy 1,502 1,323 UBS Securities Australia Ltd Buy 1,2502 1,323 UBS Securities Australia Ltd Buy 2,1,228 2,244 UBS Securities Australia Ltd Buy 1,1,227 1,188 UBS Securities Australia Ltd Buy 4,290 4,5 UBS Securities Australia Ltd Buy 4,290 4,5 UBS Securities Australia Ltd Buy 1,1,227 1,1,6 UBS Securities Australia Ltd Buy 4,290 4,5 UBS Securities Australia Ltd Buy 1,5,394 1,6,7 UBS Securities Australia Ltd Buy 2,1,539 2,631 UBS Securities Australia Ltd Buy 1,427 1,51 UBS Securities Australia Ltd Buy 2,1,539 2,685 UBS Securities Australia Ltd Buy 2,1,539 2,675 UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Buy	7,511	7992	Ordinary
UBS Securities Australia Ltd Buy 2.5332 247 UBS Securities Australia Ltd Buy 2.633 247 UBS Securities Australia Ltd Buy 1,2,502 1,323 UBS Securities Australia Ltd Buy 1,731 183 UBS Securities Australia Ltd Buy 1,1,227 1,188 UBS Securities Australia Ltd Buy 1,418 150 UBS Securities Australia Ltd Buy 1,919 2,51 UBS Securities Australia Ltd Buy 1,919 2,031 UBS Securities Australia Ltd Buy 1,527 1,527 UBS Securities Australia Ltd Buy 1,527 1,527 UBS Securities Australia Ltd Buy 1,2,27 1,51 UBS Securities Australia Ltd Buy 1,2,27 1,51 UBS Securities Australia Ltd Buy 2,1,53 2,53 UBS Securities Australia Ltd Buy 2,1,53 2,53 UBS Securities Australia Ltd Buy 1,427 1,51 UBS Securities Australia Ltd Bu	-May-14	UBS Securities Australia Ltd	Buy	18,965	2,009	Ordinary
UBS Securities Australia Ltd Buy 2.637 279 UBS Securities Australia Ltd Buy 12,502 1,323 UBS Securities Australia Ltd Buy 1,722 1,244 UBS Securities Australia Ltd Buy 1,722 1,188 UBS Securities Australia Ltd Buy 1,722 1,188 UBS Securities Australia Ltd Buy 1,93 4,54 UBS Securities Australia Ltd Buy 1,93 2,031 UBS Securities Australia Ltd Buy 1,534 1,629 UBS Securities Australia Ltd Buy 1,673 1,73 UBS Securities Australia Ltd Buy 1,673 1,67 UBS Securities Australia Ltd Buy 1,67 1,67 UBS Securities Australia Ltd Buy 1,427 1,51 UBS Securities Australia Ltd Buy 1,677 1,57 UBS Securities Australia Ltd Buy 1,677 1,676 UBS Securities Australia Ltd Buy 1,678 1,676 UBS Securities Australia Ltd Buy <td>-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>2,332</td> <td>247</td> <td>Ordinary</td>	-May-14	UBS Securities Australia Ltd	Buy	2,332	247	Ordinary
UBS Securities Australia Ltd Buy 12,502 1,333 UBS Securities Australia Ltd Buy 1,731 183 UBS Securities Australia Ltd Buy 1,1,227 1,188 UBS Securities Australia Ltd Buy 4,290 454 UBS Securities Australia Ltd Buy 4,290 454 UBS Securities Australia Ltd Buy 19,193 2,031 UBS Securities Australia Ltd Buy 15,394 1,629 UBS Securities Australia Ltd Buy 22,538 2,385 UBS Securities Australia Ltd Buy 1,427 15 UBS Securities Australia Ltd Buy 1,427 15 UBS Securities Australia Ltd Buy 2,538 2,885 UBS Securities Australia Ltd Buy 1,427 15 UBS Securities Australia Ltd Buy 6,72 71 UBS Securities Australia Ltd Buy 1,5855 1,676 UBS Securities Australia Ltd Buy 1,5855 1,676 UBS Securities Australia Ltd Buy	-May-14	UBS Securities Australia Ltd	Buy	2,637	279	Ordinary
UBS Securities Australia Ltd Buy 1,731 183 UBS Securities Australia Ltd Buy 21,228 1,224 UBS Securities Australia Ltd Buy 1,427 1,188 UBS Securities Australia Ltd Buy 1,418 150 UBS Securities Australia Ltd Buy 1,913 2,031 UBS Securities Australia Ltd Buy 1,913 2,031 UBS Securities Australia Ltd Buy 1,529 2,031 UBS Securities Australia Ltd Buy 2,534 1,629 UBS Securities Australia Ltd Buy 2,253 2,629 UBS Securities Australia Ltd Buy 2,153 2,629 UBS Securities Australia Ltd Buy 2,153 2,731 UBS Securities Australia Ltd Buy 1,627 1,676 UBS Securities Australia Ltd Buy 1,825 1,676 UBS Securities Australia Ltd Buy 1,825 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd	-May-14	UBS Securities Australia Ltd	Buy	12,502	1,323	Ordinary
UBS Securities Australia Ltd Buy 21,228 2,244 UBS Securities Australia Ltd Buy 11,227 1,188 UBS Securities Australia Ltd Buy 4,290 4,54 UBS Securities Australia Ltd Buy 1,91 2 UBS Securities Australia Ltd Buy 19,193 2,031 UBS Securities Australia Ltd Buy 15,394 1,539 UBS Securities Australia Ltd Buy 22,538 2,385 UBS Securities Australia Ltd Buy 1,427 151 UBS Securities Australia Ltd Buy 21,538 2,585 UBS Securities Australia Ltd Buy 1,539 2,585 UBS Securities Australia Ltd Buy 1,585 1,676 UBS Securities Australia Ltd Buy 1,585 1,676 UBS Securities Australia Ltd Buy </td <td>5-May-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>1,731</td> <td>183</td> <td>Ordinary</td>	5-May-14	UBS Securities Australia Ltd	Buy	1,731	183	Ordinary
UBS Securities Australia Ltd Buy 11,227 1,188 UBS Securities Australia Ltd Buy 1,418 150 UBS Securities Australia Ltd Buy 4,290 454 UBS Securities Australia Ltd Buy 19,193 2,031 UBS Securities Australia Ltd Buy 15,394 1,629 UBS Securities Australia Ltd Buy 22,538 2,631 UBS Securities Australia Ltd Buy 14,27 15 UBS Securities Australia Ltd Buy 21,539 2,285 UBS Securities Australia Ltd Buy 6,72 71 UBS Securities Australia Ltd Buy 6,72 71 UBS Securities Australia Ltd Buy 1,895 1,676 UBS Securities Australia Ltd Buy 2,501 2,501	-May-14	UBS Securities Australia Ltd	Buy	21,228	2.244	Ordinary
UBS Securities Australia Ltd Buy 1,418 150 UBS Securities Australia Ltd Buy 4,290 454 UBS Securities Australia Ltd Buy 19,33 2,031 UBS Securities Australia Ltd Buy 15,394 1,629 UBS Securities Australia Ltd Buy 22,538 2,385 UBS Securities Australia Ltd Buy 1427 151 UBS Securities Australia Ltd Buy 21,539 2,685 UBS Securities Australia Ltd Buy 21,539 2,685 UBS Securities Australia Ltd Buy 672 77 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 2,501 2,501	-Mav-14	UBS Securities Australia Ltd	Buy	11,227	1.188	Ordinary
UBS Securities Australia Ltd Buy 4,590 454 UBS Securities Australia Ltd Buy 19 2 UBS Securities Australia Ltd Buy 19,193 2,031 UBS Securities Australia Ltd Buy 22,538 1,629 UBS Securities Australia Ltd Buy 14,27 15 UBS Securities Australia Ltd Buy 1,427 15 UBS Securities Australia Ltd Buy 21,538 2,885 UBS Securities Australia Ltd Buy 21,539 2,885 UBS Securities Australia Ltd Buy 1,527 71 UBS Securities Australia Ltd Buy 1,585 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 23,672 2,501	-May-14	LBS Securities Australia Ltd	Briv	1418	150	Ordinary
UBS Securities Australia Ltd Buy 19 2.7 UBS Securities Australia Ltd Buy 19.193 2.031 UBS Securities Australia Ltd Buy 15.394 1.6.29 UBS Securities Australia Ltd Buy 22.538 2.685 UBS Securities Australia Ltd Buy 1.4.27 15.1 UBS Securities Australia Ltd Buy 2.1.539 2.651 UBS Securities Australia Ltd Buy 2.1.537 2.51 UBS Securities Australia Ltd Buy 6.72 7.1 UBS Securities Australia Ltd Buy 1.6.76 1.6.76 UBS Securities Australia Ltd Buy 1.8555 1.6.76 UBS Securities Australia Ltd Buy 2.3673 2.501	N. 24. 1.4	I IBC Societies Australia 14d	200	000 8	200	Ordinary
Uses Securities Australia Ltd Buy 19,193 2,031 Uses Securities Australia Ltd Buy 19,193 2,031 Uses Securities Australia Ltd Buy 22,538 2,385 Uses Securities Australia Ltd Buy 14,27 15 Uses Securities Australia Ltd Buy 1,427 15 Uses Securities Australia Ltd Buy 21,593 2,285 Uses Securities Australia Ltd Buy 672 71 Uses Securities Australia Ltd Buy 15,855 1,676 Uses Securities Australia Ltd Buy 15,855 1,676 Uses Securities Australia Ltd Buy 1,835 1,676 Uses Securities Australia Ltd Buy 23,672 2,501	1	Ling Committee Australia Ltd	eu.	067,4	100	Clarialy
UBS Securities Australia Ltd Buy 15,739 2,031 UBS Securities Australia Ltd Buy 22,538 2,385 UBS Securities Australia Ltd Buy 142 15 UBS Securities Australia Ltd Buy 1,539 2,285 UBS Securities Australia Ltd Buy 21,593 2,285 UBS Securities Australia Ltd Buy 672 71 UBS Securities Australia Ltd Buy 1,5855 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 23,672 2,501	May	LIBS Securities Australia Ltd	Ana a	501.01	7 500 5	Ordinary
Up Securities Australia Ltd Buy 1,5,254 1,6,25 UBS Securities Australia Ltd Buy 1,2,534 2,368 UBS Securities Australia Ltd Buy 1,427 151 UBS Securities Australia Ltd Buy 2,1,539 2,568 UBS Securities Australia Ltd Buy 672 71 UBS Securities Australia Ltd Buy 1,5855 1,676 UBS Securities Australia Ltd Buy 1,885 1,676 UBS Securities Australia Ltd Buy 2,501 2,501	Man 14	Cop Decorres Australia Etc	on?	261,61	100,2	Clarialy
UBS Securities Australia Ltd Buy 2,585 UBS Securities Australia Ltd Buy 14.27 151 UBS Securities Australia Ltd Buy 21,593 2,285 UBS Securities Australia Ltd Buy 21,593 2,285 UBS Securities Australia Ltd Buy 71 71 UBS Securities Australia Ltd Buy 1,675 71 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 2,501 2,501	-May-14	UBS Securities Australia Ltd	Buy	15,394	1,629	Ordinary
UBS Securities Australia Ltd Buy 14.2 15 UBS Securities Australia Ltd Buy 1,47.2 151 UBS Securities Australia Ltd Buy 21,593 2,285 UBS Securities Australia Ltd Buy 672 71 UBS Securities Australia Ltd Buy 15,855 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 23,672 2,501	-May-14	UBS Securities Australia Ltd	gny	85,238	2,385	Ordinary
UBS Securities Australia Ltd Buy 14.27 151 UBS Securities Australia Ltd Buy 21.533 2,285 UBS Securities Australia Ltd Buy 67.2 71 UBS Securities Australia Ltd Buy 1,676 1,676 UBS Securities Australia Ltd Buy 1,835 1,676 UBS Securities Australia Ltd Buy 2,867 2,501	-May-14	UBS Securities Australia Ltd	Buy	142.	15	Ordinary
UBS Securities Australia Ltd Buy 21,593 2,285 UBS Securities Australia Ltd Buy 672 71 UBS Securities Australia Ltd Buy 15,855 1,676 UBS Securities Australia Ltd Buy 1,835 194 UBS Securities Australia Ltd Buy 2,501 2,501	-May-14	UBS Securities Australia Ltd	Buy.	1,427	151	Ordinary
UBS Securities Australia Ltd Buy 672 71 UBS Securities Australia Ltd Buy 15,855 1,676 UBS Securities Australia Ltd Buy 1,835 194 UBS Securities Australia Ltd Buy 23,672 2,501	-May-14	UBS Securities Australia Ltd	Buy	21,593	2,285	Ordinary
UBS Securities Australia Ltd Buy 1,676 1,676 UBS Securities Australia Ltd Buy 1,835 194 UBS Securities Australia Ltd Buy 23,672 2,501	-May-14	UBS Securities Australia Ltd	Buy	672	71	Ordinary
UBS Securities Australia Ltd Buy 1,835 194 UBS Securities Australia Ltd Buy 22,672 2,501	-May-14	UBS Securities Australia Ltd	Buv	15,855	1.676	Ordinary
UBS Securities Australia Ltd Buy 23,672 2,501	-May-14	UBS Securities Australia Ltd	Buv	1.835	194	, acciliance
Day Securities Australia tito	-May-14					Crairiary
	Tall and a second	It in the contract A contract the	a	23,677	2 501	Ordinary

28-May-14 UBS 5, 28-May			3.470		
88 88 88 88 88 88 88 88 88 88 88 88 88	UBS Securities Australia Ltd	Buy	7,187	763	Ordinary
88 88 88 88 88 88 88 88 88 88 88 88 88	UBS Securities Australia Ltd	Buy	19,612	2,082	Ordinary
98 88 98 98 98 98 98 98 98 98 98 98 98 9	ecurities Australia Ltd	Buy	7,366		Ordinary
98 88 98 98 98 98 98 98 98 98 98 98 98 9	ecurities Australia Ltd	Buy	4,767	905	Ordinary
UBS Se UBS SE UB	UBS Securities Australia Ltd	Buy	1,328	171	Ordinary
UBS SEL	UBS Securities Australia Ltd	Buy	2,930		Ordinary
S SBU S SBU S SBU	ecurities Australia Ltd	Buy	2,969		Ordinary
AS SAU	UBS Securities Australia Ltd	Buy	2,656	282	Ordinary
UBS S	UBS Securities Australia Ltd	Buv	32,970	3	Ordinary
UBSS	IBS Societies Australia Ltd	Bie	7,621	oua	Ordingo
	IRS Securities Australia 11d	Bin	7 950	VVS	Ordinary
) odi	Continue America 144	600	056,1	5 5	S S S S S S S S S S S S S S S S S S S
	ODS Securities Australia Liu	and.	14,130	OUC,-	Centary
8	ecurriles Australia Ltd	Buy	1,225	130	Ordinary
S S	UBS Securities Australia Ltd	Buy	1,884	200	Ordinary
NBS Se	UBS Securities Australia Ltd	Buy	99		Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	21,854		Ordinary
S SBO	Securities Australia Ltd	Buy	268'2		Ordinary
NBS Se	Securities Australia Ltd	Buy	4,371	464	Ordinary
UBS Se	scurities Australia Ltd	Buy	3,278		Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	1,196	127	Ordinary
Ī	UBS Securities Australia Ltd	Buy	16,899	1,794	Ordinary
28-May-14 UBS Se	UBS Securities Australia Ltd	Bov	029	69	Ordinary
1	UBS Securities Australia Ltd	Buv	1 093	116	Ordinary
I BS S	Securities Australia I td	Bliv	1121	119	Ordinary
UBS Se	Securities Australia Ltd	Buv	1112	118	Ordinary
I IRC C	IBS Securities Australia I td	8.0	24 015	2 5.49	Cardina
	Consisted Australia Ltd	650	210,42	010,4	O'CHING!
	Collines Actually Little	e d	505	3 8	O dilialy
S S	UBS Securities Australia LIG	any	08,042	1,283	Orginary
S C C	UBS Securities Australia Ltd	Buy	2,004	213	Ordinary
UBS Se	Securities Australia Ltd	Buy	1,722	183	Ordinary
UBS Se	scurities Australia Ltd	Buy	49,130	5,221	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	4,639	493	Ordinary
NBS Se	UBS Securities Australia Ltd	Buy	433	46	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	5,712	209	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	1,835	195	Ordinary
I IBS S	Securities Australia 11d	Biv	14 962	1 590	Ordinary
I IBC C	IBC Constitute Australia 14d	8.5	7616	900	Ordinan
S SAI	URC Securities Australia 14d	8111	7 500	707	Codings
200	USO Countries Australia I+d	(52)	000,7	100	Coloniary
000	CUITIES Australia Ltd	buy.	2,430	200	Ordinary
200	OBS Securities Australia Ltd	ing	081,/	20/	Ordinary
82	UBS Securities Australia Ltd	Buy	3,708	394	Ordinary
UBS Se	curities Australia Ltd	Buy	7,824	831	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	612	65	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	12,673	1,346	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	31,663	3,363	Ordinary
UBS Se	UBS Securities Australia Ltd	Buv	23.707	2.518	Ordinary
I IRC C	orities Australia I to	Buy	8229	715	Ordinary
100	Securities Australia Ltd	650	67/0	0.7	Con in its
	contres Australia Ltd	ânâ	505,0	04/	Ordinary
S	UBS Securities Australia Ltd	Buy	(/2	200	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	39,361	4,174	Ordinary
UBS Se	curities Australia Ltd	Buy	6,384	229	Ordinary
UBS Se	curities Australia Ltd	Buy	23,993	2,543	Ordinary
UBS Se	JBS Securities Australia Ltd	Buy	5,821	617	Ordinary
UBS Se	UBS Securities Australia Ltd	Buv	21,861	2,317	Ordinary
UBS Se	UBS Securities Australia Ltd	Buv	4.850	514	Ordinary
I IBC C	IBS Securities Australia I to	Bin	2908	OEO	Ordinan
	Cocuring Australia Ind	620	23 173	2530	Commission
con .	cultiles Australia Liu	an)	571,50	010,0	Cicillary
S C S	JBS Securities Australia Ltd	Buy	36,155	3,832	Ordinary
UBS Se	JBS Securities Australia Ltd	Buy	38	4	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	18,823	1,995	Ordinary
es san,	curities Australia Ltd	Buy	151	16	Ordinary
UBS Se	Securities Australia Ltd	Buy	32.618	3,459	Ordinary
I IRC	Servicios Australia 11d	Birk	33 044	3 505	Ordinary
200	collines Australia Lid	boy	440,55	000,0	Colliary
UBS Se	JBS Securities Australia Ltd	Buy	19,832	2,102	Ordinary
27.7	COD Securities Australia Liu	, and	200'61	2,100	Clairal y

UBS Wealth Management Australia Ltd UBS Wealth Management Australi	D KAnn 14	THE WALLET MANAGEMENT CONTROL TO	Sel	11,255	(1,197)	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	I IRS Wealth Management Australia (1d	Seil	11 210	1,1,1	Ordings
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Sell	12 910	(1 373)	Ordinary
USS Wealth Management Australia Ltd USS Wealth Management Australi	9-May-14	I IRC Wealth Management Australia I to	Coll	13.380	(C/C,1)	Ordinary
1958 Wealth Management Australia Ltd 1968 Wealth	D May 14	I IDC Month Management Activities	100	005,01	(1,423)	Cicinaly
UBS Weath Management Australia Ltd UBS Weath Manage	D May 14	I Do Woolff Management Australia Ltd	nac J	3,780	(407)	Ordinary
UBS Weath Management Australia Ltd UBS Weath Manage	O May 14	LIDO Woolth Management Australia Ltd	nac J	/000'0	(343)	Ordinary
1955 Wealth Management Australia Ltd 1965 Wealth	O May 14	I IBC Month Management Australia 140	100	155,9	(000)	000
UES Wealth Management Austraila Ltd UES Wealth Management Austrail	Q.May.14	1 BC Wealth Management Australia 1+4	buy B. II.	0500	5,070	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	I RC Wealth Management Australia 14d	in a	11 251	1 100	Oldinary
UES Wealth Management Australia Ltd UES Wealth Management Australi	O-May-14	I BC Woolth Management Augustic Ltd	200	102,11	(445)	ordinal y
UES Wealth Management Australia Ltd UES Wealth Management Australi	O-May 14	I DC Worlsh Management Australia Ltd	Jac G	4,104	(445)	Ordinary
USE Wealth Management Australia Ltd USE Wealth Management Australi	O May 14	HOC World Management Australia Ltd	buy.	0.00	/6	Ordinary
USB Wealth Management Australia Ltd USB Wealth Management Australi	9-Ividy-14	Des Wealth Management Australia Ltd	Buy	14,581	055,1	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	3-May-14	UBS Wealth Management Australia Ltd	Buy	28,533	3,033	Ordinary
UBS Weath Management Australia Ltd UBS Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Buy	5,362	570	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	13,095	1,392	Ordinary
UBS Weath Management Australia Ltd UBS Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Sell	7,212	(767)	Ordinary
UBS Weath Management Australia Ltd UBS Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Buy	3,659	389	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	7,168	762	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	8,633	1,024	Ordinary
USS Wealth Management Australia Ltd USS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	11,392	1,211	Ordinary
UES Wealth Management Australia Ltd UES Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	25,456	2,706	Ordinary
UES Wealth Management Australia Ltd UES Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	7,488	1967	Ordinary
UBS Weath Management Australia Ltd UBS Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Buy	8,514	905	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	20,311	2.159	Ordinary
UBS Weath Management Australia Ltd UBS Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Sell	10.728	(1,141)	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buv	43.453	4.619	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Seli	7.522	(800)	Ordinary
UBS Wealth Management Australia Ltd UBS Weelth Management Australi	9-May-14	UBS Wealth Management Australia (1d	Suv	11 477	1 220	Ordinary
UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell <td>9-May-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Sell</td> <td>3.347</td> <td>(356)</td> <td>Ordinary</td>	9-May-14	UBS Wealth Management Australia Ltd	Sell	3.347	(356)	Ordinary
UBS Wealth Management Austraila Ltd UBS Wealth Management Austrail	9-May-14	UBS Wealth Management Australia Ltd	Buy	14 431	1 534	Ordinary
UES Wealth Nanagement Austraie Ltd Buy UES Wealth Management Austraie Ltd Buy UES Wealth Management Austraie Ltd UES Weal	9-May-14	I IBS Wealth Management Australia Ltd	Buy	Ctour	204	Critical
USE Wealth Nanagement Australia Ltd USE Wealth Nanagement Australi	9-May-14	I IRS Wealth Management Australia Ltd	Bury	0,010	403	Ordinary
USS Wealth Management Australia Ltd USS Wealth Management Australi	O Mary 14	ODD Wealth Wallagellieft Australia tito	ond.	9,721	776	Ordinary
USS Wealth Management Australia Ltd USS Wealth Management Australi	9-IVIdy-14	Des wealth Management Australia Ltd	Buy	6,82U	57/	Ordinary
USS Weath Management Australia Ltd USS Weath Manage	S-Iviay-14	Obs Wealth Ivial lage Hear Abstralla Ltd	puy	1,18/	/78',	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	- Inlay	COS Wealth Mailagetherit Australia Ltd	and a	715,16	5,529	Ordinary
USS Wealth Management Australia Ltd USS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	968,426	102,943	Ordinary
UBS Weath Management Australia Ltd UBS Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Sell	3,046	(324)	Ordinary
USB Weath Management Australia Ltd USB Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Buy	9,295	988	Ordinary
UBS Weath Management Australia Ltd Sell Sell Sell Sell Seath Management Australia Ltd Sell Sell Sell Sell Sell Sell Sell Sel	9-May-14	UBS Wealth Management Australia Ltd	Sell	2,642	(281)	Ordinary
UES Weath Management Australia Ltd UES Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Buy	11,872	1,262	Ordinary
UES Weath Management Australia Ltd UES weath Management Australia Ltd UES Weath Manage	9-May-14	UBS Wealth Management Australia Ltd	Sell	2,971	(316)	Ordinary
USS Weath Management Australia Ltd USS Weath Management Australia Ltd	9-May-14	UBS Wealth Management Australia Ltd	Sell	3,893	(414)	Ordinary
UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Buy	9-May-14	UBS Wealth Management Australia Ltd	Buy	20,414	2,170	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australi	9-May-14	UBS Wealth Management Australia Ltd	Buy	2,880	625	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd	9-May-14	UBS Wealth Management Australia Ltd	Buy	18,702	1,988	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd	9-May-14	UBS Wealth Management Australia Ltd	Buy	19,003	2,020	Ordinary
UBS Wealth Management Australia Ltd UBS Weahth Management Australia Ltd UBS Weath Mana	9-May-14	UBS Wealth Management Australia Ltd	Buv	36.143	3.842	Ordinary
UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Buy UBS Weath Management Australia Ltd Buy UBS Weath Management Australia Ltd Buy UBS Weath Management Australia Ltd Buy UBS Weath Management Australia Ltd Buy UBS Weath Management Australia Ltd Buy UBS Weath Management Australia Ltd Buy UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell UBS Weath Management Australia Ltd Sell	9-May-14	UBS Wealth Management Australia Ltd	Buy	14,177	1,507	Ordinary
UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Suy UBS Wealth Management Australia Ltd Suy UBS Wealth Management Australia Ltd Suy UBS Wealth Management Australia Ltd Suy UBS Wealth Management Australia Ltd Suy UBS Wealth Management Australia Ltd Suy UBS Wealth Management Australia Ltd Suy UBS Wealth Management Australia Ltd Suy UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd	9-May-14	UBS Wealth Management Australia Ltd	Buy	76,078	8,087	Ordinary
UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell	9-May-14	UBS Wealth Management Australia Ltd	Sell	6.431	(684)	Ordinary
UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Managem	9-May-14	UBS Wealth Management Australia Ltd	Sell	2,379	(253)	Ordinary
UBS Wealth Management Australia Ltd Sell Sell Sell Sell Sell Sell Sell Sel	9-May-14	UBS Wealth Management Australia Ltd	Seil	13.314	(1.416)	Ordinary
UES Wealth Management Australia Ltd UES wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd	9-May-14	UBS Wealth Management Australia Ltd	Sell	3.676	(391)	Ordinary
UBS Wealth Management Australia Ltd Sell Sell UBS Wealth Management Australia Ltd Sell Sell Sell Management Australia Ltd Sell Sell Sell Sell Management Australia Ltd Sell UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell UBS Wealth Management Australia Ltd Sell Sell Sell Management Australia Ltd Sell Sell Sell Sell Sell Sell Sell Sel	9-Mav-14	UBS Wealth Management Australia Ltd	J	2 952	(314)	Ordinary
UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd UES Wealth Management Australia Ltd Sell	9-Mav-14	UBS Wealth Management Australia Ltd	Buv	15.513	1649	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd Self	9-May-14	UBS Wealth Management Australia Ltd	Sell	2,106	(224)	Ordinary
UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Self	9-May-14	UBS Wealth Management Australia Ltd	Buv	21.129	2.246	Ordinary
UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd Sell	9-May-14	UBS Wealth Management Australia Ltd	Sell	12.063	(1.283)	Ordinary
UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd Self	9-May-14	UBS Wealth Management Australia Ltd	Buv	26,510	2.818	Ordinary
UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Sell	9-May-14	UBS Wealth Management Australia Ltd	Buy	14,563	1,548	Ordinary
UBS Wealth Management Australia Ltd Buy UBS Wealth Management Australia Ltd Sell	9-May-14	UBS Wealth Management Australia Ltd	Buy	6,275	299	Ordinary
UBS Wealth Management Australia Ltd Sell	9-May-14	UBS Wealth Management Australia Ltd	Buv	069.6	1.030	Ordinary
	9-May-14	UBS Wealth Management Australia Ltd	Sell	13.878	(1,476)	Ordinary
ILIRS Wealth Management Australia Itd	9-May-14	I IRS Wealth Management Australia Ltd	Cell	7635	(010)	Ordinan,
25-May-14 Ubs wealth Management Australia Ltd Seil Seil (635)	9-May-14	U65 Wealth Management Australia Ltu	Sell	ccq'/	(812)	Ordinary

	The second second			2201	
28-May-14	UBS Securities Australia Ltd	Buy	15,834	1,672	Ordinary
20-IMdy-14	UBS Securities Australia Ltd	and a	191	-	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	9,272	977	Ordinary
8-May-14	UBS Securities Australia Ltd	Buy	14,102	1,486	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	18,458	1,945	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	15,061	1,587	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	7,753	817	Ordinary
28-May-14	UBS Securities Australia Ltd	Buv	16.171	1 704	Ordinary
8-Mav-14	UBS Securities Australia Ltd	Buv	6.728	209	Ordinary
28-May-14	UBS Securities Australia Ltd	Buv	23,513	2 475	Ordinary
28-May-14	I IRS Securities Australia (td	Bir	19 931	200.6	Ordinan
28-May 14	LIBC Countition Australia 14d	Berry	1 435	2,030	o o o
20 May 14	TIDE Consider Assets and Education	in a	CC+'-	101	Coultary
o-ividy-14	Obs Securities Australia Ltd	, and	12,490	415,1	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	188	88	Ordinary
8-May-14	UBS Securities Australia Ltd	Buy	1,113	117	Ordinary
8-May-14	UBS Securities Australia Ltd	Buy	1,484	156	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	958	06	Ordinary
28-May-14	UBS Securities Australia Ltd	Buv	4.888	514	Ordinary
28-Mav-14	UBS Securities Australia Ltd	Buv	686	104	Ordinary
8-May-14	LIBS Securities Australia I td	Riv	2 853	000	Ordinan
Name of	I DC Counting Australia 14d	600	2,000	1000	Ordinary
20-IVIdy-14	Coop Securities Australia Liu	and .	3,140	351	Ordinary
8-May-14	UBS Securities Australia Ltd	Buy	6,153	64/	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	1,060	1,163	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	1,929	202	Ordinary
8-May-14	UBS Securities Australia Ltd	Buy	353	37	Ordinary
8-May-14	UBS Securities Australia Ltd	Buy	2,101	220	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	879	35	Ordinary
8-May-14	UBS Securities Australia Ltd	Buy	47,750	2,000	Ordinary
28-May-14	UBS Securities Australia Ltd	Buy	497	52	Ordinary
8-May-14	UBS Securities Australia Ltd	BUV	4,775	200	Ordinary
8-May-14	LIRS Securities Australia Ltd	RIW	4 145	757	Ordinary
28-May-14	I IBC Countition Ametralia Ltd	Bin	77.750	000	Ordingo
D 1424 14	Let Clearant Academic Co.	654	710 61	2007	(in in
20-Ividy-14	ODD Securities Abstralia Ltd	eog.	13,21	1,304	Ciginary -:
0-ividy- (4	ODS Securities Australia Liu	Jac J	906,1	(501)	Ordinary
9-Way-14	UBS AG LONGON BRAIKEN) Sell	535,451	(2/6,95)	Ordinary
9-Ividy-14	OBS AG LOTIGOTI Brankin	gag	250,050	7/6'00	Ordinary
S-IMay-14		E, Xe	95,390	(10,143)	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	392,770	41,753	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	164,623	17,526	Ordinary
9-May-14	UBS Securities Australia Ltd	Sell	18,069	(1,924)	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	46,950	2,000	Ordinary
9-Mav-14	UBS Securities Australia Ltd	Buv	41,778	4 450	Ordinary
1 / ACA4 0	DO Continue Australia to	loo o	636 451	(0.50 27)	Ordinary Ordinary
9-May-14	UBS Securities Australia Ltd	Sell	335,451	(56,972)	Ordinary
9-May-14	UBS Securities Australia Ltd	ie.	46,900	(2,000)	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	836,569	89,100	Ordinary
9-May-14	UBS Securities Australia Ltd	Sell	1,677,022	(178,200)	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,674,920	178,200	Ordinary
9-May-14	UBS Securities Australia Ltd	Sell	419,006	(44,550)	Ordinary
29-May-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	466	Options
9-May-14	UBS Securities Australia Ltd	Enter Into Derivative C	AN	(1,319)	Options
9-May-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	800	Options
9-May-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(1,271)	Options
9-May-14	UBS Securities Australia Ltd	Enter Into Derivative (ΝΑ	(603)	Options
9-Mav-14	UBS Securities Australia Ltd	Enter Into Derivative d	N/A	(1.357)	Options
9-May-14	ILIBS Segurities Australia I td	Foter Into Derivative C	AVN	130	Ontions
94-May-14	ILIBS Sequities Australia ! td	Forter Into Derivative C	N/A	(884)	Ontions
O-May-14	I IRC Societies Australia I to	Enter Into Darinstino	VIII	(1000)	Choise
20 May 14	I IDS Commission Americal and	Cotor Into Derivative	V214	(000)	Options
2-Iviay-14	Opp Securities Australia Ltd	Eliter IIIO Delivative	(2)	(057)	Childris
9-Ividy-14	Obs securities Australia Ltd	Eller mio Delivative	42	243	Cottons
y-May-14	UBS Securities Australia Ltd	Enter into Derivative C	N/A	364	Options
9-May-14	UBS Securities Australia Ltd	Enter Into Derivative (Δ/N	(101)	Options
9-May-14	UBS Wealth Management Australia Ltd	Buy	1,167	124	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Buy	12,709	1,351	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	les	3 752	1000/	
			10.10	(666)	Orginary
29-May-14	UBS Wealth Management Australia Ltd	Sell	7.494	(767)	Ordinary

9-IMdy-14	Adamagaaat			X .	CECION
29-May-14	I IRS Wealth Management Australia Ind	Buy	963 01	000	Cidinary
29-May-14	UBS Wealth Management Australia I to	Pliv	10000	1961	Ordinary
94-May-14	I IRS Wealth Management Australia 14d	Biss	22,000	200,-	O'dilay
14 May 14	I IRC Mealth Management Australia 1+d		C/1/27	4,537	Odliay
NAME OF	I DC World Management Australia	S. S.	13,321	1,410	Ordinary
ti dela Ca	ODS Wealth Management Australia Ltd	Buy	15,541	759,1	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Buy	36,651	3,896	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	14,149	1,504	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	30,057	3,195	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Sell	3.747	(368)	Ordinary
29-May-14	II iRC Wealth Management Australia 11d	i di	14 262	1000	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1 100 Months Management A catalla to	100	2024	016,1	Cranary
52-Ivlay-14	Upo vvediti ivialiagement Australia Ltd	Sell Sell	795'/	(/83)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Seli	5,1711	(220)	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Sell	1,345	(143)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	11,364	1,208	Ordinary
99-May-14	UBS Wealth Management Australia Ltd	Bliv	4 120	438	or district
94-May-14	I IRS Wealth Management Australia Ltd	, a	7300	000	in in in
No. 14	TIDE Month Management Australia Ltd	, ind	0,004	000	Ordinary
3-Ividy-14	Obs Weakin Management Australia Ltd	Buy	3,558	75/	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	34,393	3,656	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Buy	5,964	634	Ordinary
39-May-14	UBS Wealth Management Australia Ltd	Buy	8.739	626	Ordinary
29-Mav-14	UBS Wealth Management Australia Ltd	Birv	171 897	18 272	Ordinan
29-May-14	I IRS Wealth Management Australia I to	Birk	20,00	2,202	Collings
May 14	Life Woolth Management Americal	, and	051.00	1,53	Courie
S ividy 14	Coo Wealth Managernerit Australia Ltd	eny	915,52	2,500	Crainary
9-May-14	UBS Securities Australia Ltd	Buy	195,730	23,000	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,187,375	135,700	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	631.601	70.100	Ordinan
94-May-14	I IRS Securities Australia 14d	Benz	457 500	000	
1 1 1 1	Libror Citi A strail- tal	, and	457,500	20,000	Crainary
9-Ividy-14	ODS Securities Australia Ltd	Buy	751,400	88,400	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,143,900	127,100	Ordinary
9-May-14	UBS Securities Australia Ltd	Sell	720,000	(80,000)	Ordinary
9-Mav-14	UBS Securities Australia Ltd	≥ 6	1 220 075	121 900	Codingo
9-May-14	I IRS Societies Australia I to	600	211	200,00	
	2	600		000	Cidinaly
3-May-14	Ups securities Australia Ltd	Sell	സ	(142)	Ordinary
y-May-14	UBS Securities Australia Ltd	Buy	26,762	2,844	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	301	32	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	229	72	Ordinary
29-May-14	UBS Securities Australia Ltd	Buv	592	63	Ordinary
9-May-14	ILIBS Specifities Australia 11d	i i	717	77	Ording
AL VICTOR	LIDO Cognision Australia 144	600	# 001	1	Cidillaly
9-Ividy-14	Obs securities Australia Ltd	Buy	208	54	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	470	20	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	470	20	Ordinary
29-Mav-14	UBS Securities Australia Ltd	Biv	7 858	936	Ordinary
9-May-14	I BS Securities Australia I to	Rus	1 200	127	digi
	COO OCCUPIES AUSTRALA	pag.	1,200	/21	Oramary
9-May-14	UBS Securities Australia Ltd	Buy	367	39	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	896	103	Ordinary
29-May-14	UBS Securities Australia Ltd	Buv	1.166	124	Ordinary
9-May-14	II IRS Securities Australia I to	Birk	73 550	1 636	Codingo
D NAME 14	I IDC Constitute Assets in 144		COCIO	Spir	100
ti kaini	COO Decorring Academia Etc	And	970	oc.	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	17,663	1,879	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	48,899	5,202	Ordinary
29-May-14	UBS Securities Australia Ltd	Buy	9,353	366	Ordinary
9-May-14	UBS Securities Australia Ltd	Buv	8,056	857	Ordinary
Q.May.14	I IBC Congrition Australia I to	6	793		
The Man	COO Securities Australia Liu	âna	2004	00 5	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	26,695	2,840	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,034	110	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	1,006	107	Ordinary
9-May-14	UBS Securities Australia Ltd	9nv	55.874	5.944	Ordinary
29-May-14	UBS Securities Australia I td	Biry	8 526	200	Ordinary
145,14	I In Committee Australia Ind	, ony	720,00	170	Ordinary
y-May-14	UBS Securities Australia Ltd	Buy	13,997	1,489	Ordinary
9-May-14	UBS Securities Australia Ltd	Buy	20,558	2,187	Ordinary
29-May-14	UBS Securities Australia Ltd	Buv	3.497	372	Ordinary
9-Mav-14	I IRS Securities Australia 1 to	Buy	3 760	000	Cripic
	COO COUNTRIES COOK AND COOK AN	, and	09/5	3	Oldinary 1
9-May-14	UBS Securities Australia Ltd	Buy	6.025	641	Ordinary
				5	7
29-May-14	UBS Securities Australia Ltd	Buv	12.107	1 288	Ordinary

29-May-14	I IRS Wealth Management Australia Ltd	1/g/	17 847	(1570)	Ordinan
29-May-14	UBS Wealth Management Australia Ltd	Sell	8.086	(860)	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Sell	12,007	(772,1)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	2,164	230	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Sell	12.609	(1.341)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Sell	6,751	(718)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	10,941	1,163	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Buy	13,377	1,422	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Selt	7,710	(820)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Seli	13,502	(1,436)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	13,330	1.417	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buv	5.259	559	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	7.422	789	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Sell	14,348	(1,526)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	BCV	17.648	1.876	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	62.039	6,599	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Sel	9,393	(666)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buv	46.219	4.913	Ordinary
-Mav-14	UBS Wealth Management Australia Ltd	Buy	79 991	3 188	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	12.314	1.309	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Sell	10,174	(1,082)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	21.872	2,325	Ordinary
29-May-14	UBS Wealth Management Australia Ltd) Se	7.494	(262)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buv	22,051	2.344	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Buv	16,717	1.777	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	23,989	2,550	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Sell	480	(51)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	865	92	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buv	12.201	1.297	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Buy	13,932	1.481	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Sell	8.519	(906)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	10,517	1,118	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	23,914	2,542	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Buv	33.707	3.583	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Sell	15,411	(1,639)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Sell	6,535	(695)	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Buy	11,016	1,171	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Sell	13,474	(1.433)	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Sell	4,128	(439)	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Sell	5,153	(548)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Sell	9,365	(966)	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Sell	5,557	(163)	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Buy	24,205	2,573	Ordinary
-May-14	UBS Wealth Management Australia Ltd	Sell	2,492	(565)	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	8cv	9,972	1.060	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buv	12.023	1.278	Ordinary
9-Mav-14	UBS Wealth Management Australia Ltd	9	8 970	(P56)	Ordinary
Mav-14	I IRS Wealth Management Australia Ltd	* is	25.184	7677	Ordinan
29-May-14	I IRS Wealth Management Australia 1td	(a)	11 810	(1.256)	Ordinary
9-May-14	I IRC Wealth Management Australia I to	2 2	27 035	2 /38	Ordinan
20 May 14	DISC Month Management Australia Ltd	600	44,530 50 COA	2,430	Ordinary
29-May-14	II IRS Wealth Management Australia (14)	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	05 PC	3 122	Ordinary
9-May-14	I IRC Wealth Management Australia Ltd	log S	12,600	7,142	Ordinary
March	Hige Woolfth Management Australia Ind		2,003	(140,1)	Croinary
20-May-14	Libe Weater Management Australia Ltd	And And	45,300	4,000	Ordinary
29-May-14	URC Wealth Management Australia Ltd	, a	12,000	1 205	Ordinary
May 14	Libe Wealth Managerier Adsualia Ltd	ng and	12,009	1,405	Ordinary
9-May-14	LIDS Wealth Management Australia Ltd	and d	13,144	4,035	Ordinary
May 14	UBS Wealth Management Australia Ltd	had	G65'071	13,457	Ordinary
May-14	UBS wealth Management Australia Ltd	Buy	1,733	822	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Sell	11,772	(1,252)	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	17,686	1,880	Ordinary
29-May-14	UBS Wealth Management Australia Ltd	Buy	15,645	1,663	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Buy	10,075	1,071	Ordinary
9-May-14	UBS Wealth Management Australia Ltd	Buy	16,265	1 720	Ordinary
				(77)	5
29-May-14	IUBS Wealth Management Australia Ltd	Buy	12,681	1,348	Ordinary

V * * * * * O	UBS Sequrities Australia I to	Buv	6,237	099	Ordinary
30-ivlay-14					
30 May 14	UBS Securities Australia Ltd	euy	10,045	1,063 1	Ordinary
30-May-14	UBS Securities Australia Ltd	Buy	4,725	200	Ordinary
30-May-14	UBS Securities Australia Ltd	Buy	10,508	1,112	Ordinary
30-May-14	UBS Securities Australia Ltd	Buy	123	13	Ordinary
30-May-14	UBS Securities Australia Ltd	Buv	841	08	Ordinary
30-May-14	UBS Securities Australia Ltd	Buv	99	7	Ordinary
30-May-14	It IRC Securities Australia Ltd	Bus	35 560	222.0	in in
O Mary 14	I IOC Constitute Average at the		200,00	500	in in in
20 May 14	CDD Securities Australia Etc	bug	12,574	755,1	Orginary
O-INIdy-14	ODS Securities Australia Lita	pny	42,235	4,4,4	Ordinary
30-May-14	UBS Securities Australia Ltd	Buy	4,673	495	Ordinary
30-May-14	UBS Securities Australia Ltd	Buy	1,114	118	Ordinary
30-May-14	UBS Securities Australia Ltd	Buy	4,493	476	Ordinary
30-May-14	UBS Securities Australia Ltd	Buy	5,645	298	Ordinary
30-May-14	UBS Securities Australia Ltd	Seil	191 024	(20.257)	Ordinary
02-Jun-14	UBS AG London Branch	Buv	1 015 061	108 546	Ordinary
02-Jun-14	UBS AG London Branch	lay.	1 021 092	(108 546)	Ording
02-1-17	I IRC Con without Australia 144	3	260,120,1	(000)	Oldinary
72 him 16	I IDC Contration Australia Ltd	1130	500'0	(607)	Ordinary
1 1 1	ODD Securities Australia Ltd	(ng	100,001	10,085	Ordinary
1 1 1 1 2	Opp Securities Australia Liu	Jan C	000,88	(000,01)	Orginary
	ODS Securities Australia Lid	Buy	108,934	11,600	Ordinary
22-Jun-14	UBS Securities Australia Ltd	Buy	113,495	12,052	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Sell	244,517	(22,908)	Ordinary
32-Jun-14	UBS Securities Australia Ltd	Sell	1,021,092	(108,546)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Self	6,238	(663)	Ordinary
32-Jun-14	UBS Securities Australia Ltd	Buy	98,208	10,427	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Self	278.425	(29.631)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (ΨN	(250)	Ontions
2-Jun-14	UBS Securities Australia Ltd	Finter loto Derivative (Δ/N	2,0	Ontions
2-hm-14	LIRC Woolth Management Australia 14d	200	190.7	(1.65)	Coptions
	The second secon		CO2'/!	(0.00,1)	Cidinary
2-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,266	(1,833)	Ordinary
2-Jun-14	UBS Wealth Management Australia Ltd	Setl	17,323	(1,839)	Ordinary
2-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,473	(1,218)	Ordinary
2-Jun-14	UBS Wealth Management Australia Ltd	Setl	5,765	(612)	Ordinary
02-Jun-14	UBS Wealth Management Australia Ltd	Buy	4,237	450	Ordinary
2-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,511	(1,222)	Ordinary
12-Jun-14	UBS Wealth Management Australia Ltd	las	12.029	(1.277)	Ordinary
7-lin-14	HiRS Wealth Management Australia 11d	100	292.3	(613)	Ordinan
12-110-14	I IBC Meath Macacoment Australia Ital	loo!	11 012	(4.00.4)	S I
	COO WESTININGHAGETHEN COST SHE ELO	100	11,012	(1,234)	Ordinary
72-Jun-14	UBS Wearth Management Australia Ltd	Seli	18,2/4	(1,940)	Ordinary
72-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,681	(1,877)	Ordinary
2-Jun-14	UBS Wealth Management Australia Ltd	Seli	11,765	(1,249)	Ordinary
2-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,511	(1,222)	Ordinary
02-Jun-14	UBS Wealth Management Australia Ltd	Seli	18,943	(2,011)	Ordinary
2-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,229	(1829)	Ordinary
2-4-n-14	IRC Wealth Management Australia 114		17.364	(5,042)	
- 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	ODS Wealth Ivial agentient Australia Eth	IIAC	105,11	(1,843)	Ordinary
02-1011-14	Obs Wealth Management Australia Ltd	Sell	17,464	(1,854)	Ordinary
02-JUN-14	UBS Wealth Management Australia Ltd	Seil	11,520	(1,223)	Ordinary
02-Jun-14	UBS Wealth Management Australia Ltd	Seil	17,116	(1,817)	Ordinary
02-Jun-14	UBS Wealth Management Australia Ltd	Sell	18,077	(1,919)	Ordinary
02-Jun-14	UBS Wealth Management Australia Ltd	Sell	16,127	(1,712)	Ordinary
02-Jun-14	UBS Wealth Management Australia Ltd	Sell	6.613	(702)	Ordinary
02-Jun-14	UBS Wealth Management Australia Ltd	les	11 577	(1 229)	Ordinary
12-hin-14	I BS Wealth Adapagement Australia I to	log	11 520	(1 223)	Ordinary
17	Libo Woolfth Management Australia Ltd	190	730 21	(4,023)	S C
20.20	ODS Wealth Management Australia Ltd	Jac I	/67'/	(200,1)	Orginary
2-3UD-14	UBS Wealth Management Australia Ltd	Sell	/57'/1	(1,832)	Ordinary
72-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,483	(1,219)	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	90,871	9,616	Ordinary
32-Jun-14	UBS Securities Australia Ltd	Buy	1,162	123	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Suv	54.413	5 758	Ordinary
02-lun-14	I BC Countition Australia 14d	0	17 904	1 00 4	o district
	ODD DECARITIES MUSICIAINE ELIC	ng)	100,71	+00'-	Gurlary
72-JUN-14	UBS Securities Australia Ltd	Seil	53,595	(3,55)	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Sell	13,391	(1,417)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	9	00000		
		3	48,484	(5,184)	Ordinary
02-Jun-14	UBS Securities Australia Ltd	3,4	48,989	(5,184)	Ordinary

URS Securities Autorials Lud Byp 1,259 1,250 URS Securities Autorials Lud Byp 1,259 1,250 URS Securities Autorials Lud Byp 7,271 1,258 URS Securities Autorials Lud Byp 7,271 1,0346 URS Securities Autorials Lud Byp 4,262 5,00 URS Securities Autorials Lud Byp 5,00 2,82 URS Securities Autorials Lud Byp 7,50 2,82 URS Securities Autorials Lud Byp 7,50 2,80 URS Securities Autorials Lud Byp	* * *		,,,,	2.350	250	Ordinary
BS Securines Autorials Led Buy 7,129 1,126 1,261 1,062 1	ay-14	UBS Securities Australia Ltd	and.		257	(al all al
Observative Autorials Led Bluy 97,273 98,80 Observative Autorials Led Bluy 97,273 10,348 Observative Autorials Led Bluy 4,526 580 Observative Autorials Led Bluy 3,760 460	ay-14	UBS Securties Australia Ltd	Buy	11,806	1,256	Ordinary
Resources Autorials Ltd Bluy 97,271 10,284 Bis Securities Autorials Ltd Bluy 4,326 5,24 5,24 1,227	ay-14	UBS Securities Australia Ltd	Buy	7,219	768	Ordinary
UBS Securities Autoratial Ltd Bluy 4,226 524 UBS Securities Autoratial Ltd Bluy 4,226 524 UBS Securities Autoratial Ltd Bluy 4,782 562 UBS Securities Autoratial Ltd Bluy 4,782 662 UBS Securities Autoratial Ltd Bluy 2,500 266 UBS Securities Autoratial Ltd Bluy 2,500 266 UBS Securities Autoratial Ltd Bluy 3,760 400 UBS Securities Autoratial Ltd	lay-14	UBS Securities Australia Ltd	Buy	97,271	10,348	Ordinary
49.00 49.00 49.00 56.92 58.0 49.00 49.00 54.00 58.0 58.0 49.00 49.00 54.00 58.0 68.0 49.00 49.00 49.00 58.0 68.0 40.00 80.00 49.00 49.00 68.0 40.00 80.00 49.00 49.00 49.00 40.00 80.00 80.00 49.00 49.00 40.00 80.00 80.00 7.240 80.00 40.00 80.00 80.00 7.240 80.00 40.00 80.00 80.00 7.240 80.00 40.00 80.00 80.00 7.240 80.00 40.00 80.00 80.00 7.240 80.00 40.00 80.00 80.00 80.00 80.00 40.00 80.00 80.00 80.00 80.00 40.00 80.00 80.00 80.00 80.00 40.00 <t< td=""><td>lay-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>11,534</td><td>1,227</td><td>Ordinary</td></t<>	lay-14	UBS Securities Australia Ltd	Buy	11,534	1,227	Ordinary
UBS Securities Autarials Ltd Buy 5,425 SEQ UBS Securities Autarials Ltd Buy 4,702 5,62 SEQ UBS Securities Autarials Ltd Buy 4,702 4,62 6,60 UBS Securities Autarials Ltd Buy 2,500 2,60 2,60 4,	lay-14	UBS Securities Australia Ltd	Buy	4,926	524	Ordinary
UBS Securities Autarials Ltd Buy 4,700 550 UBS Securities Autarials Ltd Buy 5,228 464 UBS Securities Autarials Ltd Buy 2,500 4,500 469 UBS Securities Autarials Ltd Buy 2,500 4,500 469 UBS Securities Autarials Ltd Buy 7,448 803 400 UBS Securities Autarials Ltd Buy 3,700 7,248 803 UBS Securities Autarials Ltd Buy 3,700 3,700 400 UBS Securities Autarials Ltd Buy 3,700 3,700 400 UBS Securities Autarials Ltd Buy 5,700 5,700 400 UBS Securities Autarials Ltd Buy 5,700 5,800 10,33 UBS Securities Autarials Ltd Buy 5,700	1ay-14	UBS Securities Australia Ltd	Buy	5,452	280	Ordinary
UBS Securities Australia Ltd Bly 9,033 9,62 LDS Securities Australia Ltd Bly 4,320 1,620 LDS Securities Australia Ltd Bly 4,320 1,620 LDS Securities Australia Ltd Bly 7,700 2,500 2,650 LDS Securities Australia Ltd Bly 7,700 2,600 2,600 2,600 LDS Securities Australia Ltd Bly 7,700 7,700 2,600 2,600 LDS Securities Australia Ltd Bly 7,700 7,700 2,600 2,600 LDS Securities Australia Ltd Bly 2,600 2,600 2,600 2,600 LDS Securities Australia Ltd Bly 2,600 2,600 2,600 2,600 LDS Securities Australia Ltd Bly 2,600 2,600 2,600 2,600 LDS Securities Australia Ltd Bly 2,600 2,600 2,600 2,600 LDS Securities Australia Ltd Bly 2,600 2,600 2,600 2,600 2,600 2,600 2,600 <th< td=""><td>lav-14</td><td>UBS Securities Australia Ltd</td><td>Buv</td><td>4.700</td><td>200</td><td>Ordinary</td></th<>	lav-14	UBS Securities Australia Ltd	Buv	4.700	200	Ordinary
UBS Securities Australia Ltd Buy 1,5228 1,520 UBS Securities Australia Ltd Buy 2,500 4,200 UBS Securities Australia Ltd Buy 2,500 4,000 UBS Securities Australia Ltd Buy 3,700 3,800 UBS Securities Australia Ltd Buy 5,900 4,500 UBS Securities Australia Ltd Sell Sell 1,500 4,500 UBS Securities Australia Ltd Sell Sell 1,500 1,500 500 UBS Securities Australia Ltd Sell Sell 1,500 1,500 1,500 UBS Securities Australia Ltd Sell Sell 1,500 1,500 1,500 UBS Securities Australia Ltd Sell Sell 1,500 1,500 1,500 UBS Securities Australia Ltd Sell Sell 1,500 1,500 1,500 UBS Securities Australia Ltd Sell Sell 1,500	lav-14	UBS Securities Australia Ltd	Buv	9.043	696	Ordinary
UBS Securities Australia Ltd Buy 4,262 464 LDBS Securities Australia Ltd Buy 2,500 2,60 466 100 2,500 2,500 2,60 400 100 2,500 2,60 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 400 100 2,50 100 2,50 100 2,50 100 2,50 100 2,50 100 100 2,50 100 100 2,50 100 100 2,50 100 2,50 100 2,50 100 2,50 100 2,50 100 2,50 100 2,50 100 2,50 100 2,50 100 2,50 100 2,50 100	lav-14	I IBS Seminities Australia Ltd	Briv	15,728	1,620	Ordinary
Obs. Securities Australia Lud Buy 2,526 646 Obs. Securities Australia Lud Buy 2,520 687 DBS Securities Australia Lud Buy 7,548 7,548 DBS Securities Australia Lud Buy 3,750 3,939 UBS Securities Australia Lud Buy 3,750 3,939 UBS Securities Australia Lud Buy 3,750 3,939 UBS Securities Australia Lud Buy 3,750 3,939 UBS Securities Australia Lud Buy 3,750 4,00 UBS Securities Australia Lud Buy 5,400 4,00 UBS Securities Australia Lud Buy 5,500 1,00 UBS Securities Australia Lud Buy <td>14</td> <td>Laboration American</td> <td>60</td> <td>C3C 4</td> <td>950</td> <td>O'dinary</td>	14	Laboration American	60	C3C 4	950	O'dinary
UBS Securities Australia Ltd Buy 4,200 6,00 UBS Securities Australia Ltd Buy 7,700 4,00 UBS Securities Australia Ltd Buy 7,70 3,70 4,00 UBS Securities Australia Ltd Buy 7,70 3,70 3,70 1,	dy 14	ODD DECURIES AUSTRAIN LIG	pank	4,502	\$ 2	Ordinary
UBS Securities Australia Ltd	19/-14	Ubs securities Australia Ltd	BUJ	2,500	997	Ordinary
UBS Securities Australia Ltd Buy 3,760 400 UBS Securities Australia Ltd Buy 7,540 807 UBS Securities Australia Ltd Buy 7,542 803 UBS Securities Australia Ltd Buy 7,547 3,939 UBS Securities Australia Ltd Buy 5,622 5,63 UBS Securities Australia Ltd Buy 2,617 2,078 UBS Securities Australia Ltd Buy 2,617 2,078 UBS Securities Australia Ltd Buy 2,617 2,08 UBS Securities Australia Ltd Buy 6,532 6,80 UBS Securities Australia Ltd Sell 8,20 6,532 6,80 UBS Securities Australia Ltd Sell 8,20 1,300 1,00 UBS Securities Australia Ltd Sell 8,20 1,300 1,00 UBS Securities Australia Ltd Buy 2,50 1,300 1,00 UBS Securities Australia Ltd Buy 2,00 1,300 1,300 UBS Securities Australia Ltd Buy 2,00 <td>1ay-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>780</td> <td>83</td> <td>Ordinary</td>	1ay-14	UBS Securities Australia Ltd	Buy	780	83	Ordinary
UBS Securities Australia Ltd BMy 7,548 80.9 UBS Securities Australia Ltd Buy 37,27 3,939 UBS Securities Australia Ltd Buy 5,672 5,69 UBS Securities Australia Ltd Buy 5,672 5,60 UBS Securities Australia Ltd Buy 5,677 20,763 UBS Securities Australia Ltd Buy 5,670 400 UBS Securities Australia Ltd Buy 5,670 400 UBS Securities Australia Ltd Buy 5,707 2,890 UBS Securities Australia Ltd Sell 2,800 6,539 6,639 UBS Securities Australia Ltd Sell 8,810 8,500 6,530 1,000 UBS Securities Australia Ltd Sell 8,810 8,500 1,000 1,000 UBS Securities Australia Ltd Sell 8,810 1,000 1,000 1,000 UBS Securities Australia Ltd Sell 8,810 1,000 1,000 1,000 UBS Securities Australia Ltd Sell 8,910 1,000	lay-14	UBS Securities Australia Ltd	Buy	3,760	400	Ordinary
UBS Securities Australia Ltd Buy 37027 3,939 UBS Securities Australia Ltd Buy 5,622 5,63 UBS Securities Australia Ltd Buy 5,632 5,63 UBS Securities Australia Ltd Buy 20,013 2,129 UBS Securities Australia Ltd Buy 20,013 2,129 UBS Securities Australia Ltd Buy 20,013 2,129 UBS Securities Australia Ltd Buy 2,613 2,620 UBS Securities Australia Ltd Sell 2,710,72 2,880 UBS Securities Australia Ltd Sell 2,50 1,693 UBS Securities Australia Ltd Sell 2,50 1,600 UBS Securities Australia Ltd Sell 3,50 1,633 UBS Securities Australia Ltd Sell 8,50 1,623 UBS Securities Australia Ltd Sell 8,50 1,623 UBS Securities Australia Ltd Buy 2,682 1,623 UBS Securities Australia Ltd Buy 2,682 1,623 UBS Securities Australia Ltd	lay-14	UBS Securities Australia Ltd	Buy	7,548	803	Ordinary
UBS Securities Australia Ltd Buy 5472 558 UBS Securities Australia Ltd Buy 5457.72 20.763 UBS Securities Australia Ltd Buy 5.620.33 20.763 UBS Securities Australia Ltd Buy 6.523 20.763 UBS Securities Australia Ltd Buy 6.523 400 UBS Securities Australia Ltd Buy 6.523 6.529 UBS Securities Australia Ltd Buy 2.7072 2.880 UBS Securities Australia Ltd Sell 3.756 (13) UBS Securities Australia Ltd Sell 3.756 (14) UBS Securities Australia Ltd	1ay-14	UBS Securities Australia Ltd	Buy	37,027	3,939	Ordinary
UBS Securities Australia Ltd Buy 5,452 580 UBS Securities Australia Ltd Buy 20,713 20,763 UBS Securities Australia Ltd Buy 20,013 21,729 UBS Securities Australia Ltd Buy 20,013 21,732 680 UBS Securities Australia Ltd Buy 3,760 4,00 UBS Securities Australia Ltd Sell 27,022 2,880 UBS Securities Australia Ltd Sell 27,022 2,880 UBS Securities Australia Ltd Sell 2,522 6,830 UBS Securities Australia Ltd Sell 9,256 (1,333) UBS Securities Australia Ltd Sell 9,260 (5,800) UBS Securities Australia Ltd Sell 9,260 (1,324) UBS Securities Australia Ltd Buy 2,049 (1,324) UBS Securities Australia Ltd Sell 1,033 UBS Securities Australia Ltd Sell 2,045 (1,324) UBS Securities Australia Ltd Sell 2,045 (1,324) UBS Securities Aust	1ay-14	UBS Securities Australia Ltd	Buy	517	55	Ordinary
UBS Securities Australia Ltd Buy 25,172 20,763 UBS Securities Australia Ltd Buy 6,301 2,129 UBS Securities Australia Ltd Buy 6,332 6,800 UBS Securities Australia Ltd Buy 6,333 6,93 UBS Securities Australia Ltd Sell 2,800 4,000 UBS Securities Australia Ltd Sell 2,800 5,930 UBS Securities Australia Ltd Sell 2,800 5,930 UBS Securities Australia Ltd Sell 1,63,500 5,930 UBS Securities Australia Ltd Sell 1,63,500 5,930 UBS Securities Australia Ltd Sell 1,63,500 5,100 UBS Securities Australia Ltd Sell 1,23,224 1,4,550 UBS Securities Australia Ltd Sell 1,2,22 1,547 UBS Securities Australia Ltd Sell 2,1,30 1,550 UBS Securities Australia Ltd Buy 2,1,32 1,500 UBS Securities Australia Ltd Buy 2,1,80 1,5,50 UBS Securities	1av-14	UBS Securities Australia Ltd	Buy	5.452	580	Ordinary
UBS Securities Australia Ltd Buy 2,0,13 2,1,29 UBS Securities Australia Ltd Buy 6,332 680 UBS Securities Australia Ltd Buy 6,533 680 UBS Securities Australia Ltd Sell 2,1,29 400 UBS Securities Australia Ltd Sell 2,20 405 UBS Securities Australia Ltd Sell 2,25 6,83 UBS Securities Australia Ltd Sell 2,25 (6,83) UBS Securities Australia Ltd Sell 2,25 (7,133) UBS Securities Australia Ltd Buy 5,58,29 (7,133) UBS Securities Australia Ltd Buy 1,63,20 (7,133) UBS Securities Australia Ltd Buy 1,22,24 (4,033) UBS Securities Australia Ltd Sell 1,20,24 (4,201) UBS Securities Australia Ltd Sell 1,20,24 (1,204) UBS Securities Australia Ltd Sell 1,20,24 (1,204) UBS Securities Australia Ltd Sell 2,049,014 (1,204) UBS Securit	Aav-14	LIBS Securities Australia Itd	Bit	195 172	50.00	Ordinan
UBS Securities Australia Ltd Buy 6,392 6,80 UBS Securities Australia Ltd Buy 6,392 6,80 UBS Securities Australia Ltd Buy 6,332 6,69 UBS Securities Australia Ltd Sell 28 (3) UBS Securities Australia Ltd Buy 56,825 (5,83) UBS Securities Australia Ltd Sell 399,160 (42,407) UBS Securities Australia Ltd Buy 46,800 5,800 UBS Securities Australia Ltd Sell 399,160 (42,407) UBS Securities Australia Ltd Buy 2,087 (1,500) UBS Securities Australia Ltd Sell 39,483 (1,500) UBS Securities Australia Ltd Buy 2,087 (1,829) UBS Securities Australia Ltd Sell 8,83 (1,500) UBS Securities Australia Ltd	177.17	I IRC Countries Australia 140	P. C.	21.755	0110	Ordinary
UBS Securities Australia Ltd Buy 3,760 400 UBS Securities Australia Ltd Buy 3,760 400 UBS Securities Australia Ltd Buy 2,23 653 693 UBS Securities Australia Ltd Sell 27,012 2,890 (1) UBS Securities Australia Ltd Sell 2,20 (3) (3) UBS Securities Australia Ltd Sell 2,20 (3) UBS Securities Australia Ltd Sell 2,50 (3) UBS Securities Australia Ltd Sell 39,160 (1),333 UBS Securities Australia Ltd Buy 5,58,22 (1),433 UBS Securities Australia Ltd Buy 2,04,20 (1),330 UBS Securities Australia Ltd Buy 2,04,20 (1),20 UBS Securities Australia Ltd Buy 2,04,20 (1),20 UBS Securities Australia Ltd Buy 2,04,20 (1),20 UBS Securities Australia Ltd Sell 2,04,20 (1),20 UBS Securities Australia Ltd Sell 2,04,20 (1),2		COO SECURIOS ACTUALISTA LA	600	5050	67177	Cicinaly
UBS Securities Australia Ltd Buy 5,750 400 UBS Securities Australia Ltd Buy 5,750 2,890 UBS Securities Australia Ltd Sell 2,8 (1) UBS Securities Australia Ltd Sell 1,8,2,50 (1),333 UBS Securities Australia Ltd Buy 1,8,2,23 (1,0,000) UBS Securities Australia Ltd Buy 1,8,2,23 (1,0,000) UBS Securities Australia Ltd Sell 1,8,2,23 (1,0,000) UBS Securities Australia Ltd Sell 1,8,2,73 (1,0,000) UBS Securities Australia Ltd Buy 2,0,400 (1,0,000) UBS Securities Australia Ltd Buy 2,0,400 (1,0,000) UBS Securities Australia Ltd Sell 2,0,400 (1,0,000) UBS Securities Australia Ltd Sell 2,0,400 (1,0,000)	1dy-14	Upp Securities Australia Ltd	Buy	265'0	080	Ordinary
UBS Securities Australia Ltd Buy 6,533 695 UBS Securities Australia Ltd Sell 28 (1) UBS Securities Australia Ltd Sell 28 (1) UBS Securities Australia Ltd Sell 28 (1) UBS Securities Australia Ltd Buy 5,5829 (1) UBS Securities Australia Ltd Buy 4,6800 (1,233) UBS Securities Australia Ltd Buy 132,524 (1,000) UBS Securities Australia Ltd Sell 399,160 (42,407) UBS Securities Australia Ltd Buy 182,221 (1,000) UBS Securities Australia Ltd Buy 182,221 (1,000) UBS Securities Australia Ltd Buy 2,38 (1,000) UBS Securities Australia Ltd Buy 2,39 (1,500) UBS Securities Australia Ltd Buy 2,39 (1,500) UBS Securities Australia Ltd Sell 2,39 (1,500) UBS Securities Australia Ltd Sell 7,202 (1,500) UBS Securities Australia Ltd<	7ay-14	UBS Securities Australia Ltd	Buy	3,760	400	Ordinary
UBS Securities Australia Ltd Buy 27,072 2,80 UBS Securities Australia Ltd Sell 28 (13) UBS Securities Australia Ltd Sell 28 (13) UBS Securities Australia Ltd Sell 18,520 (18,333) UBS Securities Australia Ltd Sell 18,520 (17,333) UBS Securities Australia Ltd Buy 46,800 5,900 UBS Securities Australia Ltd Buy 18,232 (10,000) UBS Securities Australia Ltd Buy 2,1820 (15,500) UBS Securities Australia Ltd Buy 2,1820 (15,500) UBS Securities Australia Ltd Buy 2,1820 (1,524) UBS Securities Australia Ltd Sell 2,1820 (1,524) UBS Securities Australia Ltd Sell 2,2920 (1,172) UBS	1ay-14	UBS Securities Australia Ltd	Buy	6,533	695	Ordinary
UBS Securities Australia Ltd Sell 28 (1) UBS Securities Australia Ltd Sell 28 (1) UBS Securities Australia Ltd Sell 2.56.29 (1933) UBS Securities Australia Ltd Buy 5.620 (1933) UBS Securities Australia Ltd Buy 5.620 (1933) UBS Securities Australia Ltd Buy 5.620 (10,030) UBS Securities Australia Ltd Buy 112,234 (10,000) UBS Securities Australia Ltd Buy 18,232 (10,000) UBS Securities Australia Ltd Buy 18,232 (10,000) UBS Securities Australia Ltd Buy 18,232 (10,000) UBS Securities Australia Ltd Buy 2,234 (10,000) UBS Securities Australia Ltd Buy 2,0350 (1,324) UBS Securities Australia Ltd Sell 1,220 (1,324) UBS Securities Australia Ltd Sell 1,220 (1,324) UBS Securities Australia Ltd Sell 7,202 (1,324) UBS Securi	fay-14	UBS Securities Australia Ltd	Buy	27,072	2,880	Ordinary
UBS Securities Australia Ltd Sell 28 (1) UBS Securities Australia Ltd Sell 9,250 (833) UBS Securities Australia Ltd Buy 55,829 5,330 UBS Securities Australia Ltd Buy 46,800 5,330 UBS Securities Australia Ltd Buy 46,800 5,000 UBS Securities Australia Ltd Buy 46,800 5,000 UBS Securities Australia Ltd Sell 13,254 (10,000) UBS Securities Australia Ltd Buy 46,800 5,600 UBS Securities Australia Ltd Buy 10,800 5,600 UBS Securities Australia Ltd Buy 20,930 1,540 UBS Securities Australia Ltd Buy 20,930 1,520 UBS Securities Australia Ltd Buy 20,930 1,520 UBS Securities Australia Ltd Sell 2,321 1,520 UBS Securities Australia Ltd Sell 2,321 1,520 UBS Securities Australia Ltd Sell 2,321 1,520 UBS Securities Australia Ltd<	fay-14	UBS Securities Australia Ltd	Sell	28	(3)	Ordinary
UBS Securities Australia Ltd Sell 28 (3) UBS Securities Australia Ltd Buy 55,839 5,930 UBS Securities Australia Ltd Buy 55,839 5,900 UBS Securities Australia Ltd Sell 163,520 (17,333) UBS Securities Australia Ltd Sell 399,160 (42,407) UBS Securities Australia Ltd Buy 122,524 14,033 UBS Securities Australia Ltd Buy 122,524 14,033 UBS Securities Australia Ltd Buy 182,232 19,343 UBS Securities Australia Ltd Buy 182,327 1,500 UBS Securities Australia Ltd Buy 2,049,014 2,547 UBS Securities Australia Ltd Buy 2,049,014 2,547 UBS Securities Australia Ltd Sell 2,049,014 2,547 UBS Securities Australia Ltd Sell 2,049,014 2,172 UBS Securities Australia Ltd Sell 2,049,014 2,172 UBS Securities Australia Ltd Buy 2,049 2,172 <td< td=""><td>1av-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>6</td><td>Ê</td><td>Ordinary</td></td<>	1av-14	UBS Securities Australia Ltd	Sell	6	Ê	Ordinary
UBS Securities Australia Ltd Sell 9,250 (983) UBS Securities Australia Ltd Buy 5,930 (1933) UBS Securities Australia Ltd Sell 163,500 (17,000) UBS Securities Australia Ltd Buy 46,800 5,000 UBS Securities Australia Ltd Sell 39,693 (10,000) UBS Securities Australia Ltd Buy 132,224 14,010) UBS Securities Australia Ltd Buy 182,222 (10,000) UBS Securities Australia Ltd Buy 1,000 1,000 UBS Securities Australia Ltd Buy 2,040,014 2,17,580 UBS Securities Australia Ltd Buy 2,040,014 2,17,580 UBS Securities Australia Ltd Buy 2,040,014 2,17,29 UBS Securities Australia Ltd Sell 2,040,014 2,17,20 UBS Securities Australia Ltd Sell 2,040,014 2,17,20 UBS Securities Australia Ltd Sell 2,040,014 2,17,20 UBS Securities Australia Ltd Buy 2,040,014 2,17,20	14-Ve	UBS Securities Australia I td	llas	28	(2)	Ordinary
UBS Securities Australia Ltd Buy 55,829 5,930 UBS Securities Australia Ltd Sell 165,520 5,930 UBS Securities Australia Ltd Sell 165,520 5,000 UBS Securities Australia Ltd Buy 46,800 5,000 UBS Securities Australia Ltd Sell 39,639 (10,033) UBS Securities Australia Ltd Sell 39,639 (10,000) UBS Securities Australia Ltd Buy 20,637 (2,210) UBS Securities Australia Ltd Buy 2,087 (2,210) UBS Securities Australia Ltd Buy 2,089 (1,040) UBS Securities Australia Ltd Buy 2,049 (2,500) UBS Securities Australia Ltd Buy 2,049 (1,504) UBS Securities Australia Ltd Sell 1,040 (1,504) UBS Securities Australia Ltd Sell 2,271 (1,524) UBS Securities Australia Ltd Sell 1,202 (1,524) UBS Securities Australia Ltd Buy 2,271 (2,172) UBS	1av-14	I IRS Securities Australia I td	9	9 250	(580)	Ordinary
UBS Securities Australia Ltd Sell 163.50 (17.333) UBS Securities Australia Ltd Buy 46.800 5,000 UBS Securities Australia Ltd Buy 122,254 14,033 UBS Securities Australia Ltd Sell 399,160 (42,407) UBS Securities Australia Ltd Sell 20,874 (2,210) UBS Securities Australia Ltd Buy 140,801 15,000 UBS Securities Australia Ltd Buy 140,801 15,000 UBS Securities Australia Ltd Buy 2,049,014 2,1,580 UBS Securities Australia Ltd Buy 2,049,014 2,1,580 UBS Securities Australia Ltd Buy 2,049,014 2,1,580 UBS Securities Australia Ltd Sell 2,1,627 4,291 UBS Securities Australia Ltd Sell 2,1,627 (1,829) UBS Securities Australia Ltd Sell 2,7,20 (1,829) UBS Securities Australia Ltd Buy 2,7,20 (1,829) UBS Securities Australia Ltd Buy 2,7,10 1,42 1,2<	14	1 IDC Countries Australia 1+4		053,530	000 1	Ordinary
UBS Securities Australia Ltd Sell 16,320 1 (1,329) UBS Securities Australia Ltd Sell 399,160 45,000 5,000 UBS Securities Australia Ltd Sell 399,160 (42,407) UBS Securities Australia Ltd Sell 39,160 (42,407) UBS Securities Australia Ltd Buy 1,22,25 (10,000) UBS Securities Australia Ltd Buy 1,92,23 (2,10) UBS Securities Australia Ltd Buy 2,049,014 27,589 UBS Securities Australia Ltd Buy 2,049,014 27,589 UBS Securities Australia Ltd Buy 2,049,014 27,589 UBS Securities Australia Ltd Sell 2,321 (3,204) UBS Securities Australia Ltd Sell 2,321 (3,204) UBS Securities Australia Ltd Sell 2,27,92 (1,324) UBS Securities Australia Ltd Sell 2,27,92 (1,2712) UBS Securities Australia Ltd Buy 2,27,92 (2,1712) UBS Securities Australia Ltd Buy 2,27,92	1	COS Seculities Australia Ltd	ina ina	929'00'	006,c	Oldinary
UBS Securities Australia Ltd Bluy 46,800 UBS Securities Australia Ltd Sell 3693 UBS Securities Australia Ltd Buy 132,254 14,033 UBS Securities Australia Ltd Sell 20,834 (2,100) UBS Securities Australia Ltd Buy 182,232 19,343 UBS Securities Australia Ltd Buy 182,232 19,343 UBS Securities Australia Ltd Buy 182,232 19,343 UBS Securities Australia Ltd Buy 2,049,014 2,547 UBS Securities Australia Ltd Buy 2,049,014 2,547 UBS Securities Australia Ltd Sell 2,049,014 2,1559 UBS Wealth Management Australia Ltd Sell 2,049,014 2,224 UBS Wealth Management Australia Ltd Sell 2,794 (1,829) UBS Securities Australia Ltd Sell 2,794 (1,829) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 30 UBS Securities Australia Ltd	lay- 14	UBS Securities Australia Ltd	Sell	025,581	(17,333)	Ordinary
UBS Securities Australia Ltd Seil 139,160 (42,407) UBS Securities Australia Ltd Seil 132,254 14,033 UBS Securities Australia Ltd Seil 20,874 (2,210) UBS Securities Australia Ltd Buy 18,222 19,343 UBS Securities Australia Ltd Buy 14,0391 15,000 UBS Securities Australia Ltd Buy 23,989 2,547 UBS Securities Australia Ltd Buy 2,049,014 27,550 UBS Securities Australia Ltd Buy 2,049,014 27,589 UBS Securities Australia Ltd Buy 2,049,014 27,589 UBS Securities Australia Ltd Seil 2,321 4,520 UBS Wealth Management Australia Ltd Seil 2,321 4,521 UBS Securities Australia Ltd Seil 2,279 2,4172) UBS Securities Australia Ltd Buy 2,279 1,524 UBS Securities Australia Ltd Buy 2,279 2,4172 UBS Securities Australia Ltd Buy 2,279 2,61	1ay-14	UBS Securities Australia Ltd	Buy	46,800	2,000	Ordinary
UBS Securities Australia Ltd Buy 132,254 14,033 UBS Securities Australia Ltd Sell 20,693 (10,000) UBS Securities Australia Ltd Buy 182,222 19,343 UBS Securities Australia Ltd Buy 182,222 19,343 UBS Securities Australia Ltd Buy 20,390 (5,500) UBS Securities Australia Ltd Buy 23,980 2,547 UBS Securities Australia Ltd Buy 23,980 2,547 UBS Securities Australia Ltd Buy 20,901 48,317 5,61 UBS Securities Australia Ltd Sell 21,887 (1,829) UBS Securities Australia Ltd Sell 2,784 (1,829) UBS Securities Australia Ltd Sell 7,808 (1,829) UBS Securities Australia Ltd Sell 7,808 (2,171) UBS Securities Australia Ltd Buy 2,643 50 UBS Securities Australia Ltd Buy 2,643 50 UBS Securities Australia Ltd Buy 2,643 50	1ay-14	UBS Securities Australia Ltd	Sell	399,160	(42,407)	Ordinary
UBS Securities Australia Ltd Sell 29,693 (10,000) UBS Securities Australia Ltd Sell 20,874 (2,210) UBS Securities Australia Ltd Buy 140,801 15,000 UBS Securities Australia Ltd Buy 2,049,014 2,17,589 UBS Securities Australia Ltd Buy 2,049,014 2,17,589 UBS Securities Australia Ltd Buy 4,8317 4,550 UBS Securities Australia Ltd Sell 2,165 4,520 UBS Wealth Management Australia Ltd Sell 1,152 4,520 UBS Wealth Management Australia Ltd Sell 1,7202 (1,829) UBS Securities Australia Ltd Sell 2,7942 (1,829) UBS Securities Australia Ltd Sell 2,7942 (7,77) UBS Securities Australia Ltd Buy 2,7942 (7,77) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,479 2,62	lay-14	UBS Securities Australia Ltd	Buy	132,254	14,033	Ordinary
UBS Securities Australia Ltd Sell 20,874 (2,210) UBS Securities Australia Ltd Buy 182,232 19,343 UBS Securities Australia Ltd Buy 1,920 (5,500) UBS Securities Australia Ltd Buy 2,647 2,756 UBS Securities Australia Ltd Buy 2,049,014 27,559 UBS Securities Australia Ltd Buy 2,043,014 27,559 UBS Georgies Australia Ltd Sell 48,317 5,161 UBS Georgies Australia Ltd Sell 21,827 (2,324) UBS Wealth Management Australia Ltd Sell 11,202 (1,943) UBS Wealth Management Australia Ltd Sell 17,202 (1,824) UBS Securities Australia Ltd Sell 227,942 (2,41,12) UBS Securities Australia Ltd Sell 7808 (828) UBS Securities Australia Ltd Buy 2,479 1,5 UBS Securities Australia Ltd Buy 2,479 2,6 UBS Securities Australia Ltd Buy 2,479 2,6 <td< td=""><td>av-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>269 266</td><td>(000001)</td><td>Ordinary</td></td<>	av-14	UBS Securities Australia Ltd	Sell	269 266	(000001)	Ordinary
UBS Securities Australia Ltd Buy 182,232 19,343 UBS Securities Australia Ltd Buy 18,232 19,200 UBS Securities Australia Ltd Buy 2,049,014 2,547 UBS Securities Australia Ltd Buy 2,049,014 2,75,589 UBS Securities Australia Ltd Buy 2,049,014 2,75,589 UBS Securities Australia Ltd Sell 2,049,014 2,73,241 UBS Wealth Management Australia Ltd Sell 8,615 (1,829) UBS Wealth Management Australia Ltd Sell 2,792 (1,829) UBS Wealth Management Australia Ltd Sell 2,794 (1,829) UBS Wealth Management Australia Ltd Sell 2,794 (1,829) UBS Securities Australia Ltd Sell 2,794 (1,829) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 30 UBS Securities Australia Ltd Buy 2,619 27 UBS Securities Australia Ltd Buy 2,619 27	1av-14	LIBS Securities Australia Ltd	lle?	20.874	(0107)	Ordina
UBS Securities Australia Ltd Buy 16,275 UBS Securities Australia Ltd Self 51,920 (5,500) UBS Securities Australia Ltd Buy 2,647 (5,500) UBS Securities Australia Ltd Buy 2,043,014 2,7,589 UBS Securities Australia Ltd Buy 2,043,014 2,7,589 UBS Gourities Australia Ltd Self 4,321 4,550 UBS Wealth Management Australia Ltd Self 1,8,055 (1,924) UBS Wealth Management Australia Ltd Self 17,202 (1,924) UBS Securities Australia Ltd Self 2,7,942 (1,924) UBS Securities Australia Ltd Self 7,808 (8,88) UBS Securities Australia Ltd Buy 2,7,942 (7,77) UBS Securities Australia Ltd Buy 2,7942 (7,77) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,479 2,62 UBS Securities Australia Ltd Buy 2,479 2,62 UBS Securities A	10, 14	I IDC Countries Australia 14d		100,02	012/2)	Ordinary Ordinary
UBS Securities Australia Ltd Sell 14,0801 UBS Securities Australia Ltd Sell 15,920 (5,500) UBS Securities Australia Ltd Buy 2,3989 2,547 (5,600) UBS Securities Australia Ltd Buy 2,049,014 2,17,589 2,547 17,589 1,620 1,650 1,620 1,650 1	1	ODS Securities Australia Liu	áng	102,232	19,545	Cluring
UBS Securities Australia Ltd Buy 25,920 (5,500) UBS Securities Australia Ltd Buy 2,049,014 2,71,589 UBS Securities Australia Ltd Buy 2,049,014 2,71,589 UBS Securities Australia Ltd Buy 4,327 4,520 UBS Wealth Management Australia Ltd Sell 1,625 (1,324) UBS Wealth Management Australia Ltd Sell 1,7,202 (1,829) UBS Securities Australia Ltd Sell 2,7,202 (1,829) UBS Securities Australia Ltd Sell 2,7,202 (1,829) UBS Securities Australia Ltd Sell 7,202 (1,829) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,647 1,5 UBS Securities Australia Ltd Buy 2,679 2,67 UBS Securities Australia Ltd Buy 2,679 2,67 UBS Securities Australia Ltd Buy 2,679 2,62 UBS Securities Australia Ltd Buy 2,679 2,62 <th< td=""><td>ay-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>140,801</td><td>000,21</td><td>Ordinary</td></th<>	ay-14	UBS Securities Australia Ltd	Buy	140,801	000,21	Ordinary
UBS Securities Australia Ltd Buy 2,547 UBS Securities Australia Ltd Buy 2,049,014 2,1558 UBS Securities Australia Ltd Buy 48,317 5,161 UBS Good Securities Australia Ltd Buy 48,327 5,161 UBS Good Seen Management Australia Ltd Sell 21,827 (2,324) UBS Woelth Management Australia Ltd Sell 18,055 (1,924) UBS Wealth Management Australia Ltd Sell 17,202 (1,829) UBS Securities Australia Ltd Sell 227,942 (4,172) UBS Securities Australia Ltd Sell 7,808 (8,28) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,641 7,112 UBS Securities Australia Ltd Buy 2,479 2,62 UBS Securities Australia Ltd Buy 2,479 2,62 UBS Securities Australia Ltd Buy 2,479 2,62 UBS Securities Aus	lay-14	UBS Securities Australia Ltd	Sell	51,920	(5,500)	Ordinary
UBS Securities Australia Ltd Buy 2,049,014 217,559 UBS Securities Australia Ltd Buy 4,8317 5,161 UBS Global Asset Namagement Australia Ltd Sell 2,827 4,520 UBS Wealth Management Australia Ltd Sell 1,829 (1,324) UBS Wealth Management Australia Ltd Sell 1,7202 (1,829) UBS Wealth Management Australia Ltd Sell 27,942 (1,829) UBS Wealth Management Australia Ltd Sell 2,7942 (1,829) UBS Securities Australia Ltd Sell 7,266 (7,71) UBS Securities Australia Ltd Buy 1,82 1,6 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 30 UBS Securities Australia Ltd Buy 2,843 30 UBS Securities Australia Ltd Buy 7,119 7,11 UBS Securities Australia Ltd Buy 2,479 2,62 UBS Securities Australia Ltd Buy 7,119 7,11	lay-14	UBS Securities Australia Ltd	Buy	23,989	2,547	Ordinary
UBS Securities Australia Ltd Buy 48,317 5,161 UBS Global Asset Management Australia Ltd Sell 21,921 4,550 UBS Wealth Management Australia Ltd Sell 18,095 (1,924) UBS Wealth Management Australia Ltd Sell 17,202 (1,924) UBS Securities Australia Ltd Sell 227,942 (1,829) UBS Securities Australia Ltd Sell 7,808 (828) UBS Securities Australia Ltd Buy 227,942 (7,7172) UBS Securities Australia Ltd Buy 2,78 (7,7172) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 30 UBS Securities Australia Ltd Buy 2,66 27 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 3,764 400 UBS Secur	lay-14	UBS Securities Australia Ltd	Buy	2,049,014	217,589	Ordinary
UBS Global Asset Management Buy 42,921 4,550 UBS Weath Management Australia Ltd Sell 18,57 (2,324) UBS Weath Management Australia Ltd Sell 18,095 (1,924) UBS Weath Management Australia Ltd Sell 17,202 (1,829) UBS Securities Australia Ltd Sell 77,942 (1,829) UBS Securities Australia Ltd Sell 77,06 (771) UBS Securities Australia Ltd Buy 1,82 1,5 UBS Securities Australia Ltd Buy 1,12 1,5 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,119 7,21 UBS Securities Australia Ltd Buy 7,19 7,17 UBS Securities Australia Ltd Buy 1,230 1,00 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Lt	lay-14	UBS Securities Australia Ltd	Buy	48,317	5,161	Ordinary
UBS Wealth Management Australia Ltd Self 21,857 (2,324) UBS Wealth Management Australia Ltd Self 18,095 (1,924) UBS Wealth Management Australia Ltd Self 18,095 (1,924) UBS Securities Australia Ltd Self 17,202 (1,829) UBS Securities Australia Ltd Self 726 (771) UBS Securities Australia Ltd Buy 142 15 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,647 27 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 2,479 2,62 UBS Securities Australia Ltd Buy 2,479 2,62 UBS Securities Australia Ltd Buy 3,744 400 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd<	lav-14	UBS Global Asset Management	Buy	42 921	4.550	Ordinary
UBS Wealth Management Australia Ltd Sell 1,025 (1,024) UBS Wealth Management Australia Ltd Sell 1,020 (1,024) UBS Wealth Management Australia Ltd Sell 17,202 (1,029) UBS Securities Australia Ltd Sell 227,942 (4,172) UBS Securities Australia Ltd Sell 7,808 (628) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,001 950 UBS Securities Australia Ltd Buy 2,643 30 UBS Securities Australia Ltd Buy 2,66 27 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd<	17	I IBC Mosteh Management Australia Ltd	Soll	24 057	WCC C/	Ordinary Ordinary
UBS Wealth Management Australia Ltd Sell 18,049-1 UBS Wealth Management Australia Ltd Sell 18,049-1 UBS Securities Australia Ltd Sell 27,342 UBS Securities Australia Ltd Sell 726 UBS Securities Australia Ltd Sell 726 UBS Securities Australia Ltd Buy 142 UBS Securities Australia Ltd Buy 2,843 UBS Securities Australia Ltd Buy 2,843 UBS Securities Australia Ltd Buy 2,843 UBS Securities Australia Ltd Buy 7,119 UBS Securities Australia Ltd Buy 7,26 UBS Securities Australia Ltd Buy 7,596 UBS Securities Australia Ltd Buy 7,596 UBS Securities Australia Ltd Buy 3,744 UBS Securities Australia Ltd Buy 3,794 UBS Securities Australia Ltd	4 - I	Ubbs wealth Management Australia Ltd	Sell	/58,12	(2,324)	Ordinary
UBS Wealth Management Australia Ltd Sell (916) UBS Securities Australia Ltd Sell 17,202 (18,29) UBS Securities Australia Ltd Sell 227,942 (24,172) UBS Securities Australia Ltd Sell 7,808 (828) UBS Securities Australia Ltd Sell 7,60 (77) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,643 300 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3	tay-14	UBS Wealth Management Australia Ltd	Sell	18,095	(1,924)	Ordinary
UBS Wealth Management Australia Ltd Sell 17,202 (1,829) UBS Securities Australia Ltd Sell 7,808 (24,172) UBS Securities Australia Ltd Sell 7,808 (771) UBS Securities Australia Ltd Buy 1,82 1,57 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,479 2,62 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,749 400 UBS Securities Australia Ltd Buy 3,749 400 UBS Securities Australia Ltd Buy 3,749 400 UBS Securities Australia Ltd Buy<	lay-14	UBS Wealth Management Australia Ltd	Sell	8,615	(916)	Ordinary
UBS Securities Australia Ltd Sell 227,942 (24,172) UBS Securities Australia Ltd Sell 7,808 (828) UBS Securities Australia Ltd Sell 7,7 (7) UBS Securities Australia Ltd Buy 1,42 1,5 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 2,56 27 UBS Securities Australia Ltd Buy 7,199 33 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy	lay-14	UBS Wealth Management Australia Ltd	Sell	17,202	(1.829)	Ordinary
UBS Securities Australia Ltd Sell 7,808 (828) UBS Securities Australia Ltd Sell 776 (77) UBS Securities Australia Ltd Buy 1,47 (6) UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,19 7,71 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,33	lav-14	UBS Securities Australia I1d	las	727 942	(24 172)	Ordinary
UBS Securities Australia Ltd Sell 7.00 UBS Securities Australia Ltd Sell 7.0 UBS Securities Australia Ltd Buy 1.42 1.6 UBS Securities Australia Ltd Buy 2.843 300 UBS Securities Australia Ltd Buy 2.843 300 UBS Securities Australia Ltd Buy 7.119 751 UBS Securities Australia Ltd Buy 2.56 2.7 UBS Securities Australia Ltd Buy 3.12 3.3 UBS Securities Australia Ltd Buy 2.479 2.62 UBS Securities Australia Ltd Buy 2.479 2.62 UBS Securities Australia Ltd Buy 7.596 803 UBS Securities Australia Ltd Buy 7.596 803 UBS Securities Australia Ltd Buy 3.784 4.00 UBS Securities Australia Ltd Buy 3.784 4.00 UBS Securities Australia Ltd Buy 3.784 4.00 UBS Securities Australia Ltd Buy 1.400 1.48	17	OCCUPATION Author 144		2 000	(000)	Codings
UBS Securities Australia Ltd Sell 726 (7/) UBS Securities Australia Ltd Buy 142 15 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 9,001 950 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,19 751 UBS Securities Australia Ltd Buy 1,627 172 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,784	1 .	Opp Securities Australia Ltd	i c	909'/	(070)	Ordinary
UBS Securities Australia Ltd Sell 57 (6) UBS Securities Australia Ltd Buy 142 15 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 7,596 27 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,400 148 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,300 130 UBS Securities Australia Ltd Buy 1,300	4-14	Ubs securities Australia Ltd	Sell	97/	(//)	Ordinary
UBS Securities Australia Ltd Buy 142 15 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 1,23 13 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 2,627 27 UBS Securities Australia Ltd Buy 1,627 172 UBS Securities Australia Ltd Buy 1,627 172 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,330	1ay-14	UBS Securities Australia Ltd	Seli	57	(9)	Ordinary
UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 2,843 300 UBS Securities Australia Ltd Buy 123 131 UBS Securities Australia Ltd Buy 7,119 751 UBS Securities Australia Ltd Buy 2,56 27 UBS Securities Australia Ltd Buy 1,627 172 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,744 400 UBS Securities Australia Ltd Buy 3,744 400 UBS Securities Australia Ltd Buy 3,741 36 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,330 130 UBS Securities Australia Ltd Buy 1,330 130	lay-14	UBS Securities Australia Ltd	Buy	142	15	Ordinary
UBS Securities Australia Ltd Buy 9,001 950 UBS Securities Australia Ltd Buy 7,119 73 UBS Securities Australia Ltd Buy 7,119 73 UBS Securities Australia Ltd Buy 2,56 27 UBS Securities Australia Ltd Buy 1,627 172 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,18 36 UBS Securities Australia Ltd Buy 3,41 36 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,300 1,33 UBS Securities Australia Ltd Buy 1,300 1,30 UBS Securities Australia Ltd Buy 1,300 1,30	lav-14	UBS Securities Australia (1d	Bliv	7.843	300	Ordinary
UBS Securities Australia Ltd Buy 7,19 731 UBS Securities Australia Ltd Buy 7,19 731 UBS Securities Australia Ltd Buy 7,19 731 UBS Securities Australia Ltd Buy 312 37 UBS Securities Australia Ltd Buy 1,627 172 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,333 143 UBS Securities Australia Ltd Buy 1,330 130	11.7	I IBC Countries Australia (+A	Billi	100 0	010	Ordingo
UBS Securities Australia Ltd Buy 7.119 751 UBS Securities Australia Ltd Buy 7.119 751 UBS Securities Australia Ltd Buy 2.56 2.7 UBS Securities Australia Ltd Buy 1.62.7 1.72 UBS Securities Australia Ltd Buy 7.596 80.3 UBS Securities Australia Ltd Buy 7.596 80.3 UBS Securities Australia Ltd Buy 3.784 4.00 UBS Securities Australia Ltd Buy 3.784 4.00 UBS Securities Australia Ltd Buy 3.41 3.6 UBS Securities Australia Ltd Buy 1.400 1.48 UBS Securities Australia Ltd Buy 1.300 1.33 UBS Securities Australia Ltd Buy 1.230 1.30	<u>.</u>	ODS Securities Australia Cid	sons.	100,6	OCK.	Oldinary
UBS Securities Australia Ltd Buy 7.119 751 UBS Securities Australia Ltd Buy 2.56 2.7 UBS Securities Australia Ltd Buy 1,627 17.2 UBS Securities Australia Ltd Buy 1,627 17.2 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,330 133 UBS Securities Australia Ltd Buy 1,230 130	lay-14	UBS Securities Australia Ltd	Buy	123	£1	Ordinary
UBS Securities Australia Ltd Buy 256 27 UBS Securities Australia Ltd Buy 312 33 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,744 400 UBS Securities Australia Ltd Buy 3,41 36 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,303 143 UBS Securities Australia Ltd Buy 1,230 130	lay-14	UBS Securities Australia Ltd	Buy	7,119	751	Ordinary
UBS Securities Australia Ltd Buy 312 33 UBS Securities Australia Ltd Buy 1,627 172 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,330 133 UBS Securities Australia Ltd Buy 1,230 130	lay-14	UBS Securities Australia Ltd	Buy	526	27	Ordinary
UBS Securities Australia Ltd Buy 1,627 172 UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 341 36 UBS Securities Australia Ltd Buy 1400 148 UBS Securities Australia Ltd Buy 1,300 133 UBS Securities Australia Ltd Buy 1,230 130	lay-14	UBS Securities Australia Ltd	Buy	312	33	Ordinary
UBS Securities Australia Ltd Buy 2,479 262 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 341 36 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,353 143 UBS Securities Australia Ltd Buy 1,230 130	av-14	UBS Securities Australia Ltd	Bliv	1.627	172	Ordinary
URS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 3,184 400 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,330 143 UBS Securities Australia Ltd Buy 1,230 130		Con Consider Assessing the	600	2 420	100	Ordinary
URS Securities Australia Ltd Buy 7,596 803 UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 341 36 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,333 143 UBS Securities Australia Ltd Buy 1,230 130	9-14	ODS Securities Australia Ltd	buy	2,419	797	Ordinary
UBS Securities Australia Ltd Buy 3,784 400 UBS Securities Australia Ltd Buy 341 36 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,353 143 UBS Securities Australia Ltd Buy 1,230 130	ay-14	UBS Securities Australia Ltd	Buy	7,596	803	Ordinary
UBS Securities Australia Ltd Buy 341 36 UBS Securities Australia Ltd Buy 1,400 148 UBS Securities Australia Ltd Buy 1,333 143 UBS Securities Australia Ltd Buy 1,230 130	ay-14	UBS Securities Australia Ltd	Buy	3,784	400	Ordinary
UBS Securities Australia Itd Buy 1,400 148 UBS Securities Australia Itd Buy 1,353 143 UBS Securities Australia Itd Buy 1,230 130	ay-14	UBS Securities Australia Ltd	8uv	341	36	Ordinary
UBS Securities Australia Ltd Buy 1,353 143 UBS Securities Australia Ltd Buy 1,230 130	av-14	UBS Securities Australia 11d	Silv.	1.400	148	Ordinary
UBS Securities Australia Ltd Buy 1,230 130	34-14	LIBS Securities Australia Ltd	Biv	1 353	143	Ordinary
UBS Securities Australia Ltd Buy	:	and our country of the country of th	co.	1 2221)	, iii
	4-1-AP			000 1	000	

2-4 m-14	1 IDC Constition Australia 1 to	S S S	0.55	77	
2-lun-14	LIBS Securities Australia Ltd	Riv	017,6	010	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buv	12.168	300	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buv	065	23	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buv	1 320	141	au o
02-Jun-14	UBS Securities Australia Ltd	Buv	778 72	£ 130	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Bliv	10	2000	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Brox	131	115	Colonial
2-kin-14	LiBS Securities Australia Ltd	Bus	107	1 6	Ordinary
J2-Jun-14	UBS Securities Australia Ltd	Buv	5 653	409	Ordinary
02-jun-14	UBS Securities Australia Itd	Briv	0300	1 063	Ordinan
02-Jun-14	UBS Securities Australia I td	Buy	31 197	3 333	Ordinary
2-lun-14	I IRS Securities Australia (14)	Bris	6.014	964	Contrary Contrary
02-lun-14	I IRS Securities Australia 14d	Briss	2 415	770	Ordinaly
14 00	Tibe commission Authority and	Suy B	C14'7	967	Ordinary
1 2	Upo comittee Australia Lita	Buy	40,557	4,333	Ordinary
02 Jun 14	LIDS Countries Australia Ltd	sna c	/81	07	Ordinary
72-Jun-14	Ups Securities Australia Ltd	Buy	17,185	1,836	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	5,391	576	Ordinary
5-Jun-7	UBS Securities Australia Lto	Buy	4,025	430	Ordinary
2-3un-14	UBS Securities Australia Ltd	Buy	10,352	1,106	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	9,411	1,006	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	292'2	830	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	9,510	1,016	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	42.186	4.507	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	76 66	3 197	Ordinan
02-Jun-14	UBS Securities Australia Ltd	Buy	15,510	1 657	Ordinan
2-lim-14	I IBS Securities Australia 14d	Buy	3007	430	Ordinary Ordinary
14 cm	De Constitut Australia 100 Constitution	600	100.00	200	Ciclinary Ciclinary
- July - 14	OBS SECUTION AUSTRAIN LLU	, and	19,984	2,135	Ordinary
- 1-11m-20	Obs securities Australia Ltd	Buy	996'/	850	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	2,040	218	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	2,274	243	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	2,293	245	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	21,219	2,267	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buv	4.025	430	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buv	777	88	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Briv	15,837	1 692	Ordinary
2-lin-14	HBS Securities Australia 14	Bin	10,510	900 C	diam'r.
02-10-14	TIPO Contract Annual City	100	610,61	7,030	Ordinary
-1-10C-70	Obs seculties Australia Liu	ilac i	511	(71)	Ordinary
+1-inc-	ODS Securities Australia LIG	Sell	931	(66)	Ordinary
72-Jun-14	UBS Securities Australia Ltd	Seli	931	(66)	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	14,467	1,539	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	9,842	1,047	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buv	3.149	335	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	689	89	Ordinany
2-hin-14	II BS Securities Australia I to	- B	A 6.2E	201	Ordinary.
12 has 14	De Constitute Australia 14d	P	000,00	764	O Colonialy
100-10	Obs securities Australia Ltd	Buy	959,51	1,451	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	470	20	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	190	84	Ordinary
)2-Jun-14	UBS Securities Australia Ltd	Buy	33,144	3,526	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	28,388	3.020	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Suv	7 492	707	Ordinary
12-4 m-14	I IBS Cognities Australia I to	i d	20102	2 2 4 4	
1	The Country August August 14.1	ána c	50,105	117'6	Ordinary
11-106-77	Opp Securities Australia Ltd	, and	33,107	3,522	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	31,894	3,393	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	10,791	1,148	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	33,727	3,588	Ordinary
32-Jun-14	UBS Securities Australia Ltd	Buy	5,010	533	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buv	2,820	300	Ordinary
- hm-14	I IBS Societities Australia 14d	Biss	2 303	245	Ordinary
F 47 CO	I IDC Commission Accession and	600	505,2	C+7	Coullary
11-110	ODS SECURITIES AUSTRAINA LLO	buy	151,85	4,165	Ordinary
02-JUN-14	UBS Securities Australia Ltd	Buy	89,723	9,545	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	21,874	2,327	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Buy	2,397	255	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Bin	1812		
			20,1	491	Ordinary
7 July 14	I IBC Cognition Australia 14d	Bene	610,4	491	Ordinary

$\ \cdot \ $
$\overline{}$
_
_
- 1
- 1
- 1
- 1
1
ı
ı
ı
1
1
1
- 1
- 1
ı
ı
1
1
ı

UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS UBS UBS UBS UBS UBS UBS UBS UBS UBS	04 km 14	I IBC Con mitton Americal to		1170		
USS Scorrifies Australia Ltd BM 1,178 1,	04-hin-14	I IRS Securities Australia Ind	in a	611	071	Clamary
USS Securities Australia Ltd Buy 1,165 1,25 USS Securities Australia Ltd Buy 2,240 2,240 4,000 USS Securities Australia Ltd Buy 2,240 2,27 2,27 USS Securities Australia Ltd Buy 2,560 2,7 2,27 USS Securities Australia Ltd Buy 2,560 2,7 2,27 USS Securities Australia Ltd Buy 2,500 2,7 2,500 USS Securities Australia Ltd Buy 2,7 2,500 2,500 USS Securities Australia Ltd Buy 2,7 2,500 2,500 USS Securities Australia Ltd Buy 2,7 2,500 2,500 USS Securities Australia Ltd Buy 2,7 2,500 2,500 USS Securities Australia Ltd Buy 2,7 2,500 2,500 USS Securities Australia Ltd Buy 2,7 2,500 2,500 USS Securities Australia Ltd Buy 2,7 2,500 2,500 USS Securities Australia Ltd Buy 2,7 2,500 USS Securities Australia Ltd Buy 2,7 2,500 1,500 USS Securities Australia Ltd Buy 2,500 1,500 USS Securities Australia Ltd Buy 2,7 2,500 1,500 USS Securities Australia Ltd Buy 2,500 1,500 USS Securities Australia Ltd Buy 2,500 1,500 USS Securities Australia Ltd Buy 2,500 1,500 USS Securities Australia Ltd Buy 2,500 1,500 USS Securities Australia Ltd Buy 2,500 1,500 USS Securities Australia Ltd Buy 2,500 1,500 USS Securities Australia Ltd Buy 3,500 1,500 US	04-Jun-14	UBS Securities Australia Ltd	Buv	1178	126	Ordinary
USS Securities Australia Ltd Buy 2,440 4,500 USS Securities Australia Ltd Buy 2,440 4,500 USS Securities Australia Ltd Buy 2,540 4,500 USS Securities Australia Ltd Buy 2,579 5,500 USS Securities Australia Ltd Buy 2,579 1,039 USS Securities Australia Ltd Buy 2,579 1,039 USS Securities Australia Ltd Buy 2,700 2,739 USS Securities Australia Ltd Buy 2,700 2,730 USS Securities Australia Ltd Buy 2,700 2,700 USS Securities Australia Ltd Buy 3,700 3,700 USS Securities Australia Ltd Buy 3,700 3,700 USS Secur	04-Jun-14	UBS Securities Australia Itd	Bry	1 160	175	Ordinary
USS Securities Australia Ltd Buy 37,400 4,000	04-1-0	I IDC Counties Australia Ltd	500	501,1	221	Ordinary
USS Securities Australia Ltd Buy 3,701 3,701 4,003 5,004 1,004 1,005 5,004 1,005 1,004 1,005 1,004 1,004 1,005 1,004 1,005 1,004 1,005 1,004 1,005	11 10	I DO Constitute Australia Liu	song d	/00/ 20	630	Ordinary
USS Securities Australia Ltd BM 2,440 2,571 USS Securities Australia Ltd BM 2,440 2,739 USS Securities Australia Ltd BM 2,555 2,600 2,739 USS Securities Australia Ltd BM 2,500 2,739 3,000 USS Securities Australia Ltd BM 2,240 400 USS Securities Australia Ltd BM 2,782 400 USS Securities Australia Ltd BM 2,782 400 USS Securities Australia Ltd BM 3,782 3,511 USS Securities Australia Ltd BM 3,782 3,511 USS Securities Australia Ltd BM 3,782 3,501 USS Securities Australia Ltd BM 3,742 4,73 USS Securities Australi	04-1 m-14	I IRS Securities Australia 114	200	2,011	2007	Ordinary
OSS SCOUNTER AMERINAL LEG BMY 2,5,739 OSS Scounter Amerinal LEG BMY 2,6,737 UBS Scounter Amerinal LEG BMY 5,737 975 UBS Scounter Amerinal LEG BMY 5,737 975 UBS Scounter Amerinal LEG BMY 5,737 976 UBS Scounter Amerinal LEG BMY 9,718 1,039 UBS Scounter Amerinal LEG BMY 9,718 1,039 UBS Scounter Amerinal LEG BMY 9,718 1,039 UBS Scounter Amerinal LEG BMY 3,030 1,039 UBS Scounter Amerinal LEG BMY 1,049 1,039 UBS Scounter Amerinal LEG BMY 1,049 1,039 UBS Scounter Amerinal LEG BMY 1,049 1,039 UBS Scounter Amerinal LEG BMY 2,733 2,030 UBS Sco	04-1-14	1 BC Cocurities Australia 14	600	10,0	325	Cidillary
USS Scornites Australia Ltd Buy \$.555 2.75 2.75 UBS Securities Australia Ltd Buy \$.757 \$.65 \$.60 UBS Securities Australia Ltd Buy \$.715 \$.737 \$.65 UBS Securities Australia Ltd Buy \$.720 \$.60 UBS Securities Australia Ltd Buy \$.720 \$.60 UBS Securities Australia Ltd Buy \$.720 \$.70 UBS Securities Australia Ltd Buy \$.720 \$.70 UBS Securities Australia Ltd Buy \$.720 \$.70 UBS Securities Australia Ltd Buy \$.70 \$.74 <td< td=""><td>04-1 m-14</td><td>I IRS Securities Australia 14d</td><td>67.0</td><td>25.610</td><td>2 720</td><td>Ordingry</td></td<>	04-1 m-14	I IRS Securities Australia 14d	67.0	25.610	2 720	Ordingry
UBS Securities Australia LIU Buy 5,797 620 LOBS Securities Australia LIU Buy 2,824 620 LOBS Securities Australia LIU Buy 2,824 56 LOBS Securities Australia LIU Buy 2,825 30 LOBS Securities Australia LIU Buy 2,733 1,88 LOBS Securities Australia LIU Buy 2,733 1,463 LOBS Securities Australia LIU Buy 2,733 3,503 LOBS Securities Australia LIU Bu	04-hn-14	UBS Securities Australia I td	Blix	8 555	015	Ordinary
UBS Securities Australia Ltd Buy 2,055 3.00 UBS Securities Australia Ltd Buy 2,051 3.00 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 2,840 400 UBS Securities Australia Ltd Buy 2,840 400 UBS Securities Australia Ltd Buy 2,840 400 UBS Securities Australia Ltd Buy 2,173 1,788 UBS Securities Australia Ltd Buy 2,170 2,73 UBS Securities Australia Ltd Buy 2,170 2,73 UBS Securities Australia Ltd Buy 2,700 2,70 UBS Securities Australia Ltd Buy	04-hm-14	LIBS Securities Australia I to	Bury	707.3	620	Ording.
UBS Securities Australia Ltd Buy 9,15 1,099 UBS Securities Australia Ltd Buy 5,24 5,6 UBS Securities Australia Ltd Buy 2,805 30 UBS Securities Australia Ltd Buy 2,805 30 UBS Securities Australia Ltd Buy 1,089 1,186 2,181 UBS Securities Australia Ltd Buy 1,686 2,29 30 UBS Securities Australia Ltd Buy 1,686 2,29 UBS Securities Australia Ltd Buy 1,686 2,29 UBS Securities Australia Ltd Buy 6,739 3,60 UBS Securities Australia Ltd Buy 2,185 2,89 UBS Securities Australia Ltd Buy 2,185 2,183 UBS Securities Australia Ltd Buy 2,185 2,143 UBS Securities Australia Ltd Buy 2,185 2,143 UBS Securities Australia Ltd Buy 2,185 3,245 UBS Securities Australia Ltd Buy 2,185 3,245 UBS Securities	04-lun-14	URS Securities Australia I td	Bury	2,005	070	Ording)
URS Securities Australia Ltd BM 5.24 4.05 URS Securities Australia Ltd BMy 3,740 400 URS Securities Australia Ltd BMy 2,805 357 URS Securities Australia Ltd BMy 2,805 351 URS Securities Australia Ltd BMy 1,168 1,168 URS Securities Australia Ltd BMy 1,241 2,23 URS Securities Australia Ltd BMy 1,168 1,463 URS Securities Australia Ltd BMy 1,169 1,463 URS Securities Australia Ltd BMy 3,702 2,89 URS Securities Australia Ltd BMy 3,702 3,89 URS Securities Australia Ltd BMy 3,702 3,89 URS Securities Australia Ltd BMy 3,702 3,90 URS Securities Australia Ltd BMy 2,805 3,00 URS Securities Australia Ltd BMy 2,805 3,00 URS Securities Australia Ltd BMy 2,805 3,00 URS Securities Australia Ltd BMy	04-lun-14	UBS Securities Australia 11d	Bis	9715	1 030	Countary
UBS Securities Australia Ltd BMy 3,740 400 UBS Securities Australia Ltd Bby 3,740 400 UBS Securities Australia Ltd Bby 3,740 3,500 UBS Securities Australia Ltd Bby 1,089 1,186 UBS Securities Australia Ltd Bby 1,089 1,683 UBS Securities Australia Ltd Bby 1,687 1,683 UBS Securities Australia Ltd Bby 2,742 473 UBS Securities Australia Ltd Bby 2,742 473 UBS Securities Australia Ltd Bby 3,740 400 UBS Securities Australia Ltd Bby 2,805 3,740 UBS Securities Australia Ltd Bby	04-lun-14	I IRS Securities Australia 14d	S S S S S S S S S S S S S S S S S S S	57.13	6507	Ordinary
UBS Securities Australia Ltd BM 3,282 3,00 UBS Securities Australia Ltd BMy 2,00 3,00 UBS Securities Australia Ltd BMy 1,00 1,00 UBS Securities Australia Ltd BMy 1,00 1,00 UBS Securities Australia Ltd BMy 2,14 2,29 UBS Securities Australia Ltd BMy 1,60 1,463 UBS Securities Australia Ltd BMy 2,179 2,33 UBS Securities Australia Ltd BMy 4,27 2,89 UBS Securities Australia Ltd BMy 4,27 2,89 UBS Securities Australia Ltd BMy 4,27 2,89 UBS Securities Australia Ltd BMy 2,175 3,90 UBS Securities Australia Ltd BMy 2,175 3,00 UBS Securities Australia Ltd BMy 2,175 3,00 UBS Securities Australia Ltd BMy 2,08 3,00 UBS Securities Australia Ltd BMy 2,08 3,00 UBS Securities Australia Ltd BMy 2	04-lim-14	I IRS Securities Australia 14d	6.78	2 740	2	Ordinary
OFOS SCORDINGS AUSTRAIGH LIVE Buy 3,282 350 UBS Securities Australia Ltd Buy 1,089 1,186 UBS Securities Australia Ltd Buy 1,689 1,186 UBS Securities Australia Ltd Buy 2,141 2.29 UBS Securities Australia Ltd Buy 6,788 7.25 UBS Securities Australia Ltd Buy 6,788 7.25 UBS Securities Australia Ltd Buy 2,702 2.89 UBS Securities Australia Ltd Buy	04-11-0-14	I IDC Cognition Australia Ltd	Series and a serie	04/1C	004	Orginary
UBS Securities Australia Ltd BMy 1,028 1,186 UBS Securities Australia Ltd BMy 1,108 1,28 UBS Securities Australia Ltd BMy 1,108 1,138 UBS Securities Australia Ltd BMy 2,14 4,03 UBS Securities Australia Ltd BMy 1,156 1,25 UBS Securities Australia Ltd BMy 1,175 1,25 UBS Securities Australia Ltd BMy 3,740 407 UBS Securities Australia Ltd BMy 3,740 407 UBS Securities Australia Ltd BMy 3,740 407 UBS Securities Australia Ltd BMy 2,875 3,89 UBS Securities Australia Ltd BMy 2,875 3,89 UBS Securities Australia Ltd BMy 2,875 3,89 UBS Securities Australia Ltd BMy 2,815 3,89 UBS Securities Australia Ltd BMy 2,815 3,89 UBS Securities Australia Ltd BMy 2,817 1,149 UBS Securities Australia Ltd BMy	04 1 14	UBS Securities Australia Ltd	Buy	208,2	300	Ordinary
UBS Securities Australia Ltd Buy 1,169 1,18 UBS Securities Australia Ltd Buy 1,169 1,29 UBS Securities Australia Ltd Buy 1,169 1,29 UBS Securities Australia Ltd Buy 1,202 1,202 UBS Securities Australia Ltd Buy 2,702 2,803 UBS Securities Australia Ltd Buy 2,702 2,803 UBS Securities Australia Ltd Buy 2,702 2,803 UBS Securities Australia Ltd Buy 2,702 2,804 UBS Securities Australia Ltd Buy 2,702 2,804 UBS Securities Australia Ltd Buy 2,702 2,804 UBS Securities Australia Ltd Buy 2,702 2,805 UBS Securities Australia Ltd Buy 2,805 3,00 UBS Securities Australia Ltd Buy 2,805 1,40 UBS Securities Australia Ltd Buy 3,246 1,00 UBS Securities Australia Ltd Buy 3,246 1,00 UBS Securities Australia Ltd Buy<	04-Jun-14	UBS Securities Australia Ltd	Buy	3,282	150	Ordinary
UBS Securities Australia Ltd Buy 196 2.7 UBS Securities Australia Ltd Buy 2.14 2.25 UBS Securities Australia Ltd Buy 2.1785 1.463 UBS Securities Australia Ltd Buy 6.2702 2.331 UBS Securities Australia Ltd Buy 4.72 4.07 UBS Securities Australia Ltd Buy 3.740 4.03 UBS Securities Australia Ltd Buy 3.740 4.07 UBS Securities Australia Ltd Buy 3.753 3.69 UBS Securities Australia Ltd Buy 3.675 3.69 UBS Securities Australia Ltd Buy 2.785 3.60 UBS Securities Australia Ltd Buy 2.805 3.00 UBS Securities Australia Ltd Buy 2.805 3.00 UBS Securities Australia Ltd Buy 2.805 3.00 UBS Securities Australia Ltd Buy 3.26 4.00 UBS Securities Australia Ltd Buy 3.70 2.80 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	680,11	98,	Ordinary
UBS Securities Australia Ltd Buy 1,141 1,274 UBS Securities Australia Ltd Buy 1,367 1,463 UBS Securities Australia Ltd Buy 6,788 725 UBS Securities Australia Ltd Buy 2,1702 1,393 UBS Securities Australia Ltd Buy 2,702 4,733 4,73 UBS Securities Australia Ltd Buy 2,702 2,89 UBS Securities Australia Ltd Buy 2,702 2,89 UBS Securities Australia Ltd Buy 2,702 3,00 UBS Securities Australia Ltd Buy 2,703 3,00 UBS Securities Australia Ltd Buy 2,0133 2,143 UBS Securities Australia Ltd Buy 2,0134 450 UBS Securities Austral	04-Jun-14	UBS Securities Australia Ltd	Buy	95.	17	Ordinary
UBS Securities Australia Ltd Buy 6,788 7.55 UBS Securities Australia Ltd Buy 6,788 7.55 UBS Securities Australia Ltd Buy 6,788 7.55 UBS Securities Australia Ltd Buy 6,788 7.55 UBS Securities Australia Ltd Buy 2,773 3,740 4.07 UBS Securities Australia Ltd Buy 2,783 3,633 UBS Securities Australia Ltd Buy 2,783 3,633 UBS Securities Australia Ltd Buy 2,783 3,632 UBS Securities Australia Ltd Buy 2,805 3,00 UBS Securities Australia Ltd Buy 2,805 3,00 UBS Securities Australia Ltd Buy 2,805 3,00 UBS Securities Australia Ltd Buy 1,149 1,149 UBS Securities Australia Ltd Buy 1,240 1,100 UBS Securities Australia Ltd Buy 1,500 3,240 1,500 UBS Securities Australia Ltd Buy 1,500 3,240 1,500	04-Jun-14	Ubs securities Australia Ltd	Buy	2,141	575	Ordinary
UBS Securities Australia Ltd Buy 6,788 725 UBS Securities Australia Ltd Buy 1,795 1,331 UBS Securities Australia Ltd Buy 2,702 2,895 UBS Securities Australia Ltd Buy 2,702 2,895 UBS Securities Australia Ltd Buy 2,702 2,895 UBS Securities Australia Ltd Buy 3,704 400 UBS Securities Australia Ltd Buy 2,702 2,895 UBS Securities Australia Ltd Buy 3,704 4,97 UBS Securities Australia Ltd Buy 2,037 2,143 UBS Securities Australia Ltd Buy 2,035 3,246 4,00 UBS Securities Australia Ltd Buy 1,048 1,036 1,439 UBS Securities Australia Ltd Buy 2,036 3,040 4,00	04-Jun-14	UBS Securities Australia Ltd	Buy	13,679	1,463	Ordinary
UBS Securities Australia Ltd Buy 18,625 1,922 UBS Securities Australia Ltd Buy 21,739 233 UBS Securities Australia Ltd Buy 2,730 203 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 2,723 3,533 UBS Securities Australia Ltd Buy 2,723 3,633 UBS Securities Australia Ltd Buy 2,733 3,633 UBS Securities Australia Ltd Buy 2,733 3,635 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 1,686 1,149 UBS Securities Australia Ltd Buy 1,680 1,149 UBS Securities Australia Ltd Buy 3,346 1,850 UBS Securities Australia Ltd Buy 3,346 1,850 UBS Securities Australia Ltd Buy 3,346 1,850 UBS Securities Australia Ltd Buy </td <td>04-Jun-14</td> <td>UBS securities Australia Ltd</td> <td>Buy</td> <td>6,788</td> <td>726</td> <td>Ordinary</td>	04-Jun-14	UBS securities Australia Ltd	Buy	6,788	726	Ordinary
UBS Securities Australia Ltd Bby 1795 2.331 UBS Securities Australia Ltd Buy 4,1735 4,03 UBS Securities Australia Ltd Buy 3,740 403 UBS Securities Australia Ltd Buy 3,743 403 UBS Securities Australia Ltd Buy 2,753 3,503 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 1,156 1,149 UBS Securities Australia Ltd Buy 1,049 1,149 UBS Securities Australia Ltd Buy 1,049 1,149 UBS Securities Australia Ltd Buy 1,049 1,056 UBS Securities Australia Ltd Buy 1,049 1,056 UBS Securities Australia Ltd Buy 3,240 4,00 UBS Securities Australia Ltd Buy 5,563 5,56 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	18,625	1,992	Ordinary
UBS Securities Australia Ltd Buy 4.423 4.93 UBS Securities Australia Ltd Buy 3.740 400 UBS Securities Australia Ltd Buy 3.702 2.89 UBS Securities Australia Ltd Buy 2.753 3.694 UBS Securities Australia Ltd Buy 2.805 300 UBS Securities Australia Ltd Buy 2.805 300 UBS Securities Australia Ltd Buy 2.805 300 UBS Securities Australia Ltd Buy 1.0809 1.149 UBS Securities Australia Ltd Buy 1.0809 1.149 UBS Securities Australia Ltd Buy 1.0809 1.149 UBS Securities Australia Ltd Buy 1.073 2.143 UBS Securities Australia Ltd Buy 1.0743 1.149 UBS Securities Australia Ltd Buy 1.000 1.000 UBS Securities Australia Ltd Buy 5.563 5.63 UBS Securities Australia Ltd Buy 5.563 5.63 UBS Securities Australia Ltd Buy </td <td>04-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>21,795</td> <td>2,331</td> <td>Ordinary</td>	04-Jun-14	UBS Securities Australia Ltd	Buy	21,795	2,331	Ordinary
UBS Securities Australia Ltd Buy 3740 400 UBS Securities Australia Ltd Buy 27,23 299 UBS Securities Australia Ltd Buy 27,73 2,89 UBS Securities Australia Ltd Buy 2,718 3,844 UBS Securities Australia Ltd Buy 2,518 3,00 UBS Securities Australia Ltd Buy 2,035 300 UBS Securities Australia Ltd Buy 2,033 1,149 UBS Securities Australia Ltd Buy 2,033 1,149 UBS Securities Australia Ltd Buy 1,049 1,000 UBS Securities Australia Ltd Buy 1,036 1,822 UBS Securities Australia Ltd Buy 1,036 1,822 UBS Securities Australia Ltd Buy 1,036 1,822 UBS Securities Australia Ltd Buy 5,63 5,63 UBS Securities Australia Ltd Buy 3,740 4,07 UBS Securities Australia Ltd Buy 5,63 5,63 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	4,423	473	Ordinary
UBS Securities Australia Ltd Buy 36.315 289 UBS Securities Australia Ltd Buy 36.315 3.83 UBS Securities Australia Ltd Buy 32,753 3.503 UBS Securities Australia Ltd Buy 2,605 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,003 1,149 UBS Securities Australia Ltd Buy 1,043 1,825 UBS Securities Australia Ltd Buy 1,036 1,824 UBS Securities Australia Ltd Buy 5,63 1,60 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	3,740	400	Ordinary
UBS Securities Australia Ltd Buy 36,315 3,844 UBS Securities Australia Ltd Buy 32,753 3,638 UBS Securities Australia Ltd Buy 25,189 2,638 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 1,149 27 UBS Securities Australia Ltd Buy 1,0743 1,149 UBS Securities Australia Ltd Buy 4,934 1,000 UBS Securities Australia Ltd Buy 4,934 1,000 UBS Securities Australia Ltd Buy 6,779 1,250 UBS Securities Australia Ltd Buy 8,700 1,240 1,000 UBS Securities Australia Ltd Buy 8,700 1,240 1,250 UBS Securities Australia Ltd Buy 8,700 1,240 1,260 UBS Securities Australia Ltd Buy 8,700 1,240 <t< td=""><td>04-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>2,702</td><td>289</td><td>Ordinary</td></t<>	04-Jun-14	UBS Securities Australia Ltd	Buy	2,702	289	Ordinary
UBS Securities Australia Ltd Buy 25,188 3,503 UBS Securities Australia Ltd Buy 25,188 2,694 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 1,149 1,149 UBS Securities Australia Ltd Buy 1,049 1,149 UBS Securities Australia Ltd Buy 1,050 1,822 UBS Securities Australia Ltd Buy 1,050 1,822 UBS Securities Australia Ltd Buy 1,050 1,822 UBS Securities Australia Ltd Buy 5,63 5,63 UBS Securities Australia Ltd Buy 5,63 5,60 UBS Securities Australia Ltd Buy 5,63 5,60 UBS Securities Australia Ltd Buy 5,63 5,63 UBS Securities Australia Ltd Buy 5,63 5,63 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	36,315	3,884	Ordinary
UBS Securities Australia Ltd Buy 25,189 2,694 UBS Securities Australia Ltd Buy 2,189 2,694 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 1,089 1,149 UBS Securities Australia Ltd Buy 1,0743 1,149 UBS Securities Australia Ltd Buy 1,686 2,143 UBS Securities Australia Ltd Buy 1,686 2,143 UBS Securities Australia Ltd Buy 4,586 2,51 UBS Securities Australia Ltd Buy 1,606 1,822 UBS Securities Australia Ltd Buy 1,7,036 1,822 UBS Securities Australia Ltd Buy 1,7,036 1,822 UBS Securities Australia Ltd Buy 3,740 4,00 UBS Securities Australia Ltd Buy 3,740 4,00 UBS Securities Australia Ltd Buy 3,740 1,384 UBS Securities Australia Ltd Buy 3,740 1,978 UBS Securities Australia Ltd	04-Jun-14	UBS Securities Australia Ltd	Buy	32,753	3,503	Ordinary
UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 3,00 UBS Securities Australia Ltd Buy 1,149 2,143 UBS Securities Australia Ltd Buy 4,58 4,93 UBS Securities Australia Ltd Buy 4,58 4,93 UBS Securities Australia Ltd Buy 6,779 1,25 UBS Securities Australia Ltd Buy 6,779 1,25 UBS Securities Australia Ltd Buy 5,63 5,63 UBS Securities Australia Ltd Buy 5,63 2,140 4,00 UBS Securities Australia Ltd <th< td=""><td>04-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>25,189</td><td>2,694</td><td>Ordinary</td></th<>	04-Jun-14	UBS Securities Australia Ltd	Buy	25,189	2,694	Ordinary
UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 1,149 1,149 UBS Securities Australia Ltd Buy 1,043 1,149 UBS Securities Australia Ltd Buy 1,046 2,143 UBS Securities Australia Ltd Buy 1,056 1,822 UBS Securities Australia Ltd Buy 1,056 1,822 UBS Securities Australia Ltd Buy 1,036 1,822 UBS Securities Australia Ltd Buy 5,739 7,53 UBS Securities Australia Ltd Buy 5,63 5,60 UBS Securities Australia Ltd Buy 5,63 5,60 UBS Securities Australia Ltd Buy 5,63 5,63 UBS Securities Australia Ltd Buy 3,740 4,07 UBS Securities Australia Ltd Buy 5,63 2,08 UBS Securities Australia Ltd Buy 6,835 7,37 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	4,974	532	Ordinary
UBS Securities Australia Ltd Buy 1,166 UBS Securities Australia Ltd Buy 10,809 1,149 UBS Securities Australia Ltd Buy 20,037 2,143 UBS Securities Australia Ltd Buy 10,743 1,149 UBS Securities Australia Ltd Buy 458 49 UBS Securities Australia Ltd Buy 458 49 UBS Securities Australia Ltd Buy 1,000 1,822 UBS Securities Australia Ltd Buy 1,7036 1,825 UBS Securities Australia Ltd Buy 5,779 7.75 UBS Securities Australia Ltd Buy 5,63 56 UBS Securities Australia Ltd Buy 5,63 56 UBS Securities Australia Ltd Buy 5,63 56 UBS Securities Australia Ltd Buy 1,344 978 UBS Securities Australia Ltd Buy 1,344 978 UBS Securities Australia Ltd Buy 1,344 978 UBS Securities Australia Ltd Buy 1,344 9	04-Jun-14	UBS Securities Australia Ltd	Buy	2,805	300	Ordinary
UBS Securities Australia Ltd Buy 10.809 11.16 UBS Securities Australia Ltd Buy 20.037 2.143 UBS Securities Australia Ltd Buy 1.95 2.1 UBS Securities Australia Ltd Buy 4.98 4.99 UBS Securities Australia Ltd Buy 9.345 1.00 UBS Securities Australia Ltd Buy 1.7036 1.822 UBS Securities Australia Ltd Buy 6.779 7.25 UBS Securities Australia Ltd Buy 6.779 7.55 UBS Securities Australia Ltd Buy 3.306 3.600 UBS Securities Australia Ltd Buy 5.563 5.67 UBS Securities Australia Ltd Buy 5.563 5.89 6.83 UBS Securities Australia Ltd Buy 3.740 4.00 1.00 UBS Securities Australia Ltd Buy 3.740 4.00 1.00 UBS Securities Australia Ltd Buy 8.350 1.00 1.00 UBS Securities Australia Ltd Buy 1.633 1.14 <td>04-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>2,805</td> <td>300</td> <td>Ordinary</td>	04-Jun-14	UBS Securities Australia Ltd	Buy	2,805	300	Ordinary
UBS Securities Australia Ltd Buy 20.037 2,143 UBS Securities Australia Ltd Buy 10,743 1,149 UBS Securities Australia Ltd Buy 458 49 UBS Securities Australia Ltd Buy 1,600 49 UBS Securities Australia Ltd Buy 1,7036 1,822 UBS Securities Australia Ltd Buy 1,7036 1,822 UBS Securities Australia Ltd Buy 1,5016 1,606 UBS Securities Australia Ltd Buy 5,63 50 UBS Securities Australia Ltd Buy 5,63 89 UBS Securities Australia Ltd Buy 6,835 1,14 UBS Securities Australia Ltd Buy 6,835 1,14 UBS Securities Australia Ltd Buy 6,835	04-Jun-14	UBS Securities Australia Ltd	Buy	10,809	1,156	Ordinary
UBS Securities Australia Ltd Buy 10,743 1,149 UBS Securities Australia Ltd Buy 196 21 UBS Securities Australia Ltd Buy 458 49 UBS Securities Australia Ltd Buy 17,036 1,22 UBS Securities Australia Ltd Buy 6,779 725 UBS Securities Australia Ltd Buy 15,016 1,606 UBS Securities Australia Ltd Buy 15,016 1,606 UBS Securities Australia Ltd Buy 1,5016 1,606 UBS Securities Australia Ltd Buy 5,563 595 UBS Securities Australia Ltd Buy 3,440 400 UBS Securities Australia Ltd Buy 3,560 833 UBS Securities Australia Ltd Buy 3,500 834 UBS Securities Australia Ltd Buy 1,394 1,384 UBS Securities Australia Ltd Buy 1,393 2,288 UBS Securities Australia Ltd Buy 1,393 2,288 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	20,037	2,143	Ordinary
UBS Securities Australia Ltd Buy 196 21 UBS Securities Australia Ltd Buy 458 49 UBS Securities Australia Ltd Buy 1945 100 UBS Securities Australia Ltd Buy 17,036 1,822 UBS Securities Australia Ltd Buy 6,779 725 UBS Securities Australia Ltd Buy 33,660 3,600 UBS Securities Australia Ltd Buy 3,336 3,87 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 2,583 2,78 UBS Securities Australia Ltd Buy 2,390 2,87 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 1,13	04-Jun-14	UBS Securities Australia Ltd	Buy	10,743	1,149	Ordinary
UBS Securities Australia Ltd Buy 458 49 UBS Securities Australia Ltd Buy 9,345 1,000 UBS Securities Australia Ltd Buy 1,7036 1,822 UBS Securities Australia Ltd Buy 1,7036 1,822 UBS Securities Australia Ltd Buy 1,5016 1,606 UBS Securities Australia Ltd Buy 3,386 637 UBS Securities Australia Ltd Buy 5,63 600 UBS Securities Australia Ltd Buy 5,63 500 UBS Securities Australia Ltd Buy 5,63 507 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 1,973 2,11 UBS Securities Australia Ltd Buy 8,350 833 UBS Securities Australia Ltd Buy 6,835 7,17 UBS Securities Australia Ltd Buy 6,835 1,17 UBS Securities Australia Ltd Buy 6,835 1,10 UBS Securities Australia Ltd Buy <	04-Jun-14	UBS Securities Australia Ltd	Buy	196	21	Ordinary
UBS Securities Australia Ltd BN 9,345 1,000 UBS Securities Australia Ltd BN 1,9345 1,000 UBS Securities Australia Ltd BN 17,038 1,822 UBS Securities Australia Ltd Buy 1,7036 1,822 UBS Securities Australia Ltd Buy 3,1016 1,606 UBS Securities Australia Ltd Buy 5,633 5,95 UBS Securities Australia Ltd Buy 5,633 5,95 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 3,140 400 UBS Securities Australia Ltd Buy 3,140 400 UBS Securities Australia Ltd Buy 3,140 400 UBS Securities Australia Ltd Buy 1,2940 1,384 UBS Securities Australia Ltd Buy 2,1393 2,188 UBS Securities Australia Ltd Buy 1,393 2,141 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy <td>04-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Bay</td> <td>458</td> <td>40</td> <td>Ordinan</td>	04-Jun-14	UBS Securities Australia Ltd	Bay	458	40	Ordinan
UBS Securities Australia Ltd Buy 1'096 1'21 UBS Securities Australia Ltd Buy 1'036 1'822 UBS Securities Australia Ltd Buy 3779 1'25 UBS Securities Australia Ltd Buy 31,860 3,600 UBS Securities Australia Ltd Buy 3,136 1,501 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 1,573 2,11 UBS Securities Australia Ltd Buy 2,309 247 UBS Securities Australia Ltd Buy 1,573 1,134 UBS Securities Australia Ltd Buy 1,373 1,134 UBS Securities Australia Ltd Buy 1,373 1,134 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Itd	Bliv	372 6	000	Ordinary
UBS Securities Australia Ltd Buy 17,036 1,822 UBS Securities Australia Ltd Buy 6,779 375 UBS Securities Australia Ltd Buy 514 350 UBS Securities Australia Ltd Buy 3,338 337 UBS Securities Australia Ltd Buy 3,348 357 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 1,373 2,18 UBS Securities Australia Ltd Buy 1,373 2,18 UBS Securities Australia Ltd Buy 1,373 2,28 UBS Securities Australia Ltd Buy 1,393 2,28 UBS Securities Australia Ltd Buy 6,835 110 UBS Securities Australia Ltd Buy 6,134 6,65 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	196	21	Ordinary
UBS Securities Australia Ltd Buy 6,779 725 UBS Securities Australia Ltd Buy 3,600 3,600 UBS Securities Australia Ltd Buy 51,41 55 UBS Securities Australia Ltd Buy 1,5016 1,606 UBS Securities Australia Ltd Buy 5,663 595 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 9,144 978 UBS Securities Australia Ltd Buy 9,144 978 UBS Securities Australia Ltd Buy 1,2940 1,384 UBS Securities Australia Ltd Buy 1,393 2,288 UBS Securities Australia Ltd Buy 1,393 2,141 UBS Securities Australia Ltd Buy 1,393 2,288 UBS Securities Australia Ltd Buy 6,835 747 UBS Securities Australia Ltd Buy 1,393 1,00 UBS Securities Australia Ltd Buy 6,134 6,56 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buv	17.036	1822	Ordinary
UBS Securities Australia Ltd Buy 33,660 3,600 UBS Securities Australia Ltd Buy 514 55 UBS Securities Australia Ltd Buy 1,5016 1,606 UBS Securities Australia Ltd Buy 3,338 337 UBS Securities Australia Ltd Buy 5,563 58 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 1,973 2,18 UBS Securities Australia Ltd Buy 2,1393 2,28 UBS Securities Australia Ltd Buy 1,973 100 UBS Securities Australia Ltd Buy 1,134 100 UBS Securities Australia Ltd Buy 1,134 11,144 UBS Securities Australia Ltd Buy 1,144 1,134 UBS Securities Australia Ltd Buy 1,144 1,134 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Buy	6779	725	Ordinary
UBS Securities Australia Ltd Buy 514 55 UBS Securities Australia Ltd Buy 15,016 1,606 UBS Securities Australia Ltd Buy 3,338 357 UBS Securities Australia Ltd Buy 5,563 595 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 9,144 978 UBS Securities Australia Ltd Buy 1,373 2,11 UBS Securities Australia Ltd Buy 1,373 2,28 UBS Securities Australia Ltd Buy 1,373 2,28 UBS Securities Australia Ltd Buy 1,573 1,10 UBS Securities Australia Ltd Buy 6,835 1,10 UBS Securities Australia Ltd Buy 6,835 1,10 UBS Securities Australia Ltd Buy 6,134 6,65 UBS Securities Australia Ltd Buy 6,134 6,134 UBS Securities Australia Ltd Buy 2,309 1,104 UBS Securities Australia Ltd Buy	04-Jun-14	UBS Securities Australia Ltd	Birv	33,660	009 8	Ordinary
UBS Securities Australia Ltd Buy 15,016 1,606 UBS Securities Australia Ltd Buy 3,338 357 UBS Securities Australia Ltd Buy 5,563 595 UBS Securities Australia Ltd Buy 5,740 400 UBS Securities Australia Ltd Buy 9,144 978 UBS Securities Australia Ltd Buy 1,373 2,13 UBS Securities Australia Ltd Buy 1,973 2,14 UBS Securities Australia Ltd Buy 2,309 2,47 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 6,134 6,134 UBS Securities Australia Ltd Buy 1,164 1,194 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy	04-itin-14	1.85 Securities Australia 11d	Rus	717	25	Ordinary
UBS Securities Australia Ltd Buy 3,336 357 UBS Securities Australia Ltd Buy 5,663 687 UBS Securities Australia Ltd Buy 5,663 589 687 UBS Securities Australia Ltd Buy 3,740 400	04-lin-14	I IRS Societies Australia 14d	Sign of the state	15,016	1,606	Ordinary
UBS Securities Australia Ltd Buy 5.939 637 UBS Securities Australia Ltd Buy 5.63 658 653 UBS Securities Australia Ltd Buy 3.740 400 UBS Securities Australia Ltd Buy 9.144 978 UBS Securities Australia Ltd Buy 12.940 1,334 UBS Securities Australia Ltd Buy 1,973 2.288 UBS Securities Australia Ltd Buy 2,309 2.47 UBS Securities Australia Ltd Buy 2,309 2.47 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy 1,683 180 UBS Securities Australia Ltd Buy 6,134 6,65 UBS Securities Australia Ltd Buy 6,134 6,65 UBS Securities Australia Ltd Buy 2,109 6,134 UBS Securities Australia Ltd Buy 2,109 6,134 6,65 UBS Securities Aus	1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	I De Courition Australia Ltd	ed.	010,61	2,000	Ordinary
UBS Securities Australia Ltd Buy 5,563 595 UBS Securities Australia Ltd Buy 5,563 595 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 1,294 978 UBS Securities Australia Ltd Buy 1,373 211 UBS Securities Australia Ltd Buy 2,399 247 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 6,134 6,56 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 6,	04-101-14	UBS Securities Australia Ltd	gna	2,338	35/	Ordinary
UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 3,740 400 UBS Securities Australia Ltd Buy 9,144 978 UBS Securities Australia Ltd Buy 8,350 893 UBS Securities Australia Ltd Buy 2,189 2,288 UBS Securities Australia Ltd Buy 2,189 2,47 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 1,154 1,134 UBS Securities Australia Ltd Buy 1,154 1,134 UBS Securities Australia Ltd Buy 6,134 1,134 UBS Securities Australia Ltd Buy 6,134 1,134 UBS Securities Australia Ltd Buy 6,134 6,134 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy	04-741-14	U. D.C. Coch mitton Attention 244	śng	600	500	Ordinary
UBS Securities Australia Ltd Buy 9.744 978 UBS Securities Australia Ltd Buy 9.144 978 UBS Securities Australia Ltd Buy 8.350 833 UBS Securities Australia Ltd Buy 1.973 2.18 UBS Securities Australia Ltd Buy 1.973 2.28 UBS Securities Australia Ltd Buy 2.309 247 UBS Securities Australia Ltd Buy 5.309 100 UBS Securities Australia Ltd Buy 1.683 100 UBS Securities Australia Ltd Buy 6.134 6.65 UBS Securities Australia Ltd Buy 2.805 300 UBS Securities Australia Ltd Buy 2.805 300 UBS Securities Australia Ltd Buy 2.805 300 UBS Securities Australia Ltd Buy	17-17-17	I IBC Cocurities Australia 144	, and	07/2 6	260	Ordinary
UBS securities Australia Ltd Buy 12,940 1,384 UBS securities Australia Ltd Buy 8,350 833 UBS Securities Australia Ltd Buy 1,973 2,11 UBS Securities Australia Ltd Buy 2,309 247 UBS Securities Australia Ltd Buy 2,309 247 UBS Securities Australia Ltd Buy 1,683 100 UBS Securities Australia Ltd Buy 1,164 1,194 UBS Securities Australia Ltd Buy 1,164 1,194 UBS Securities Australia Ltd Buy 6,134 6,56 UBS Securities Australia Ltd Buy 6,134 6,56 UBS Securities Australia Ltd Buy 6,134 6,66 UBS Securities Australia Ltd Buy 6,134 6,66 UBS Securities Australia Ltd Buy 6,134 6,59 UBS Securities Australia Ltd Buy 5,805 6,90 UBS Securities Australia Ltd Buy 5,716 5,75 UBS Securities Australia Ltd Buy	04-lin-14	LIRS Securities Australia 11d	Bit	01/10	820	Ordinary
UBS Securities Australia Ltd Buy 6,335 138 UBS Securities Australia Ltd Buy 1,973 2.13 UBS Securities Australia Ltd Buy 2,309 2.48 UBS Securities Australia Ltd Buy 2,309 2.47 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy 6,835 100 UBS Securities Australia Ltd Buy 1,164 1,194 UBS Securities Australia Ltd Buy 6,134 521 UBS Securities Australia Ltd Buy 6,134 6,56 UBS Securities Australia Ltd Buy 6,134 6,56 UBS Securities Australia Ltd Buy 6,134 6,56 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 5,75 5,75 UBS Securities Australia Ltd Buy 5,306 5,75 UBS Securities Australia Ltd Buy <	04-lin-14	I IRS Securities Australia 14d	And B	12 040	1 204	Ordinary
UBS Securities Australia Ltd Buy 1,373 213 UBS Securities Australia Ltd Buy 1,973 213 UBS Securities Australia Ltd Buy 2,309 247 UBS Securities Australia Ltd Buy 2,309 247 UBS Securities Australia Ltd Buy 6,835 170 UBS Securities Australia Ltd Buy 6,635 100 UBS Securities Australia Ltd Buy 6,134 6,65 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 5,134 6,65 UBS Securities Australia Ltd Buy 5,376 5,75 UBS Securities Australia Ltd Buy 5,376 5,75 UBS Securities Australia Ltd Buy 5,376 5,75	04-40-17	I IBC Con printed Australia Ltd	Social Control	0.350	1,004	Ordinary
UBS Securities Australia Ltd Buy 2,379 2,81 UBS Securities Australia Ltd Buy 2,309 247 UBS Securities Australia Ltd Buy 2,309 247 UBS Securities Australia Ltd Buy 935 170 UBS Securities Australia Ltd Buy 1,683 180 UBS Securities Australia Ltd Buy 1,194 1,194 UBS Securities Australia Ltd Buy 6,134 5,56 UBS Securities Australia Ltd Buy 6,134 5,66 UBS Securities Australia Ltd Buy 6,470 692 UBS Securities Australia Ltd Buy 5,260 592 UBS Securities Australia Ltd Buy 5,376 575	04-1-1-14	I BC Societies Australia 144	And	0620	244	Ordinary
UBS Securities Australia Ltd Buy 2,1393 247 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy 6,835 731 UBS Securities Australia Ltd Buy 1,683 100 UBS Securities Australia Ltd Buy 11,164 1,194 UBS Securities Australia Ltd Buy 6,134 526 UBS Securities Australia Ltd Buy 6,134 526 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 6,470 692 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 5,376 575	14	Libe Countries Australia Ltd	, ind	C/C,1	7 200	Ordinally
UBS Securities Australia Ltd Buy 6,835 731 100 1	14 14	LIBC Congrition Australia 144	and bank	000 0	2,200	Ordinary
UBS Securities Australia Ltd Buy 0,533 731 UBS Securities Australia Ltd Buy 935 100 UBS Securities Australia Ltd Buy 11,164 1,194 UBS Securities Australia Ltd Buy 3,001 321 UBS Securities Australia Ltd Buy 6,134 656 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 6,470 692 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 5,376 575	10, 10	LIBC Countries Australia Ltd	ody o	2,309	7#7	Colmary
UBS Securities Australia Ltd Buy 1,564 1,194 UBS Securities Australia Ltd Buy 1,164 1,194 UBS Securities Australia Ltd Buy 1,164 1,194 UBS Securities Australia Ltd Buy 6,134 6,56 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 6,470 692 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 5,57 UBS Securi	V1 -101 -10	Libe Countition Australia 144	ing and	2000	190	Ordinary
UBS Securities Australia Ltd Buy 11,764 1,194 UBS Securities Australia Ltd Buy 11,764 1,194 UBS Securities Australia Ltd Buy 3,001 321 UBS Securities Australia Ltd Buy 6,134 656 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 6,470 692 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 5,376 575	04-110-14	I IRS Societies Australia 11d	àng di	1 693	3 5	Ordinary
UBS Securities Australia Ltd Buy 3,001 3,73 UBS Securities Australia Ltd Buy 6,134 656 UBS Securities Australia Ltd Buy 2,805 300 UBS Securities Australia Ltd Buy 6,470 692 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 552 59	04-lim-14	I BS Securities Australia 11d	2.00	11.16	1 100	Ordinary
UBS Securities Australia Ltd Buy 5,75 UBS Securities Austr	04-lin-14	I IRS Securities Australia 34d	\$ Page 1	100 c	1,134	Ordinary
UBS Securities Australia Ltd Buy C ₁ /13+ C ₂ /20 C ₁ /13+ C ₂ /20 C ₂ /13+ C ₂ /20 C ₂	04-10-14	LIBC Societies Australia 114	Soo a	2,001	327	Ordinary
UBS Securities Australia Ltd Buy 6,470 692 UBS Securities Australia Ltd Buy 6,470 692 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 552 59	04-501-14	Libe contribut Australia Ltd	ing a	0, 134	000	Ordinary
ODS Securities Australia Ltd Buy 5,470 592 UBS Securities Australia Ltd Buy 5,376 575 UBS Securities Australia Ltd Buy 552 59	04-3011-14	UBS Securities Australia Ltd	ong .	2,605	300	Orginary
UBS Securities Australia Ltd Buy 5,578 5,578 5,9	04-Jun-14	UBS Securities Australia Ltd	Buy	0,4/0	260	Ordinary
IUBS Securities Australia Ltd (Buy 552 59	04-300-14	UBS Securities Australia Ltd	BUY	d/5,5	5/5	Crainary
	04-Jun-14	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	_			

UBS Securities Australia Ind Buy 2.2 909 2.2 909 UBS Securities Australia Ind Buy 80.7 187 80.2 180	UBS Securities Australia Ltd	Buy	22,909	2,427	Ordinary
Units Securities Autralia Ltd Buy 86.23 3.207 Units Securities Autralia Ltd Buy 80.231 8.207 Units Securities Autralia Ltd Buy 80.237 10.82.00 Units Securities Autralia Ltd Buy 80.207 10.82.00 Boss Courties Autralia Ltd Buy 80.207 10.82.00 Boss Courties Autralia Ltd Buy 80.203 10.82.00 Boss Courties Autralia Ltd Buy 80.203 10.82.00 Boss Courties Autralia Ltd Buy 80.80 10.82.00 10.82.00 Boss Courties Autralia Ltd Buy 80.80 10.82.00 10.82.00 10.82.00 Bos Courties Autralia Ltd Soil 80.80 80.03 10.82.00 10.82.00 Bos Courties Autralia Ltd Soil 80.03 80.03 80.03 10.83.00 Bos Securities Autralia Ltd Soil 80.03 80.03 10.83.00 Bos Securities Autralia Ltd Soil 80.03 10.80.00 10.83.00 Bos Securities Autralia Ltd Soil			010.00	8.530	
USS Securities Australia Ltd Sell 25/280 (23/280) (23/28	UBS Securities Australia Ltd	Buy	C12,00		Ordinary
USS Securities Australia Ltd Buy 80,973 10,024 10,05 Securities Australia Ltd Self 17,024 17,024 10,0524 1	UBS Securities Australia Ltd	Sell	257,893	(27,367)	Ordinary
USS Securities Australia Led Bluy 19,09 USS Securities Australia Led Seli 7,390 (18,204) LESS Securities Australia Led Bluy 23,805 (15,204) LESS Securities Australia Led Bluy 20,559 (2,500) LESS Securities Australia Led Bluy 20,559 (2,518) LESS Securities Australia Led Seli 8,600 (18,204) LESS Securities Australia Led Seli 8,600 (18,204) LESS Securities Australia Led Seli 8,600 (18,204) LESS Securities Australia Led Seli 2,623 (2,181) LESS Securities Australia Led Seli 8,600 (18,204) LESS Securities Australia Led Seli 2,1910 (18,204) LESS Securities Australia Led Seli 2,1910 (18,204) LESS Securities Australia Led Bluy 1,158 (18,204) LESS Securities Australia Led Bluy 1,158 (18,204) LESS Securities Australia Led Bluy 1,158 (1,158	UBS Securities Australia Ltd	Buy	30,280		Ordinary
UBS Securities Australia Ltd Sell 17,390 (18,200 UBS Securities Australia Ltd Blay 23,475 (2,500 LBS Securities Australia Ltd Blay 23,475 (2,500 LBS Securities Australia Ltd Blay 20,530 (18,200 LBS Securities Australia Ltd Sell 20,530 (18,500 LBS Securities Australia Ltd Sell 8,600 (1,511 (19,532 LBS Securities Australia Ltd Sell 8,600 (1,524 (19,532) LBS Securities Australia Ltd Sell 8,600 (1,571 (19,630) LBS Securities Australia Ltd Sell 1,751 (19,630) LBS Securities Australia Ltd Sell 1,751 (10,632) LBS Securities Australia Ltd Sell 1,751 (10,632) LBS Securities Australia Ltd Bluy 1,751 (10,712) LBS Securities Australia Ltd Bluy 1,751 (17,712) LBS Securities Australia Ltd Bluy 1,751 1,752 1,751 LBS Securities Australia Ltd </td <td>UBS Securities Australia Ltd</td> <td>8uy</td> <td>12206</td> <td></td> <td>Ordinary</td>	UBS Securities Australia Ltd	8uy	12206		Ordinary
UBS Securities Australia Ltd Sell 278,005 C25,00 UBS Securities Australia Ltd Buy 23,475 C250 UBS Securities Australia Ltd Buy 20,587 C1,181 UBS Securities Australia Ltd Sell 80,003 C1,181 UBS Securities Australia Ltd Sell 82,003 C1,181 UBS Securities Australia Ltd Sell 82,003 C1,181 UBS Securities Australia Ltd Sell 1,370 C0,03 UBS Securities Australia Ltd Buy 1,386 C1,390 C1,348 UBS Securities Australia Ltd Buy 1,386 C1,348 C1,371 UBS Securities Australia Ltd Buy 1,786 C1,371 C1,374 UBS Securities Australia Ltd Buy 1,786 C1,371 C1,371 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd	Sell	171,997		
UBS Securities Australia Ltd Buy 92.9.65 2.00 UBS Securities Australia Ltd Buy 92.96 2.05 UBS Securities Australia Ltd Buy 21,650 2.30 UBS Securities Australia Ltd Buy 20,655 2.05 UBS Weeting Macagement Australia Ltd Sell 20,653 (8,927) UBS Securities Australia Ltd Sell 2,653 (8,927) UBS Securities Australia Ltd Sell 1,751 (16,80) UBS Securities Australia Ltd Sell 1,751 (20,90) UBS Securities Australia Ltd Sell 1,751 (20,90) UBS Securities Australia Ltd Sell 1,751 (20,90) UBS Securities Australia Ltd Bluy 89,417 (25,50) UBS Securities Australia Ltd Bluy 1,753 (6,13) UBS Securities Australia Ltd Bluy 1,753 (7,13) UBS Securities Australia Ltd Bluy 1,753 (7,13) UBS Securities Australia Ltd Bluy 1,753 (7,13) UBS	UBS Securities Australia Ltd	Sell	278,085		
UBS Securities Australia Ltd Buy 99.358 10.524 UBS Securities Australia Ltd Self 20.657 (2.180) UBS Securities Australia Ltd Self 8.60.61.23 (2.181) UBS Securities Australia Ltd Self 8.60.61.23 (2.520) UBS Securities Australia Ltd Self 8.60.61.23 (3.00) UBS Securities Australia Ltd Self 2.823 (3.00) UBS Securities Australia Ltd Buy 4.634 (4.34 UBS Securities Australia Ltd Buy 1.734 (3.00) UBS Securities Australia Ltd Buy 1.734 (3.00) UBS Securities Australia Ltd Buy 1.736 1.73 U	UBS Securities Australia Ltd	Buy	23,475		
UBS Securities Australia Ltd Bluy 21,650 2,300 LBS Wealth Management Australia Ltd Sell 20,550 2,300 LBS Wealth Management Australia Ltd Sell 8,616,133 (4,800) LBS Securities Australia Ltd Sell 8,616,133 (4,800) LBS Securities Australia Ltd Sell 1,910 (20,93) UBS Securities Australia Ltd Sell 1,910 (20,93) UBS Securities Australia Ltd Sell 1,910 (20,93) UBS Securities Australia Ltd Buy 8,81 1,910 (20,93) UBS Securities Australia Ltd Buy 1,910 (20,93) 1,910 (20,93) UBS Securities Australia Ltd Buy 1,010 1,013 (3,100,000) 1,010 UBS Securities Australia Ltd Buy 1,010 1,013 1,013 1,013 UBS Securities Australia Ltd Buy 1,010 1,013 1,010 1,013 UBS Securities Australia Ltd Buy 1,010 1,023 1,023 UBS Securities Australia	UBS Securities Australia Ltd	Buy	E9E'66		
UBS Securities Australia Ltd Sell 20,557 (2,058) UBS Wealth Management Australia Ltd Sell 65,616,173 (2,059) UBS Securities Australia Ltd Sell 8,616,173 (2,001) UBS Securities Australia Ltd Sell 1,751 (1,001) UBS Securities Australia Ltd Sell 1,751 (1,001) UBS Securities Australia Ltd Sell 1,751 (1,002) UBS Securities Australia Ltd Sell 8,917 (4,1) UBS Securities Australia Ltd Sell 8,917 (4,1) UBS AC Lordon Branch Buy 8,917 (4,1) UBS AC Lordon Branch Buy 8,917 (4,1) UBS AC Lordon Branch Buy 8,917 (4,1) UBS Securities Australia Ltd Buy 1,002 (4,1) UBS Securities Australia Ltd Buy 1,003 (4,1) UBS Securities Australia Ltd Buy 1,003 (4,1) UBS Securities Australia Ltd Buy 1,003 (4,1) UBS Securities Australia Ltd </td <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>21,650</td> <td></td> <td></td>	UBS Securities Australia Ltd	Buy	21,650		
UBS Securities Australia Ltd Sell 886,616,132 (9.68) UBS Securities Australia Ltd Sell 86,073 (9.69,073) UBS Securities Australia Ltd Sell 2,823 (1.80) UBS Securities Australia Ltd Sell 1,910 (2.03) UBS Securities Australia Ltd Bluy 2,19,033 2,3,400 UBS Securities Australia Ltd Bluy 1,82,88 1,9,21 UBS Securities Australia Ltd Bluy 1,8,43 1,0,52 UBS Securities Australia Ltd Bluy 1,8,43 1,0,53 UBS Securities Australia Ltd Bluy 1,8,53 1,2,53	UBS Wealth Management Australia Ltd	Sell	20,567		
UBS Securities Australia Ltd Sell 88.616,123 (64.020) UBS Securities Australia Ltd Sell 1,910 (203) UBS Securities Australia Ltd Sell 1,910 (203) UBS Securities Australia Ltd Sell 38 1,64 UBS Securities Australia Ltd Sell 89,412 9,550 UBS Accounties Australia Ltd Sell 219,003 19,000 UBS Securities Australia Ltd Buy 18,823 1,000 UBS Securities Australia Ltd Buy 1,169 1,175 UBS Securities Austral	UBS Wealth Management Australia Ltd	Buy	19,393		
UBS Securities Australia Ltd Sell 84,003 (8.92) UBS Securities Australia Ltd Sell 1,751 (18.00) UBS Securities Australia Ltd Sell 1,751 (18.00) UBS Securities Australia Ltd Blay 89,557 9,550 UBS Securities Australia Ltd Blay 89,557 9,550 UBS AGE London Rearch Blay 89,577 9,550 UBS AGE London Rearch Blay 89,577 9,550 UBS AGE London Rearch Blay 89,177 9,550 UBS AGE London Rearch Blay 89,178 9,550 UBS AGE London Rearch Blay 89,178 9,550 UBS AGE London Rearch Blay 1,178 1,586 UBS AGE London Rearch Blay 1,178 1,586 UBS Securities Australia Ltd Blay 1,178 1,586 UBS Securities Australia Ltd Blay 1,178 1,586 UBS Securities Australia Ltd Blay 1,178 1,586 1,578 UBS Securities Australia Ltd	UBS Securities Australia Ltd	Sell	896,616,123		
UBS Securities Australia Ltd Sell 1,781 (180) UBS Securities Australia Ltd Sell 1,781 (180) UBS Securities Australia Ltd Sell 1,781 (180) UBS Securities Australia Ltd Buy 89,412 9,550 UBS AGE London Branch Buy 89,412 2,940 UBS AGE London Branch Buy 219,013 2,3408 UBS Securities Australia Ltd Buy 1,134 1,0500 UBS Securities Australia Ltd Buy 1,156 1,178 1,25 UBS Securities Australia Ltd Buy 1,169 1,178 1,25 UBS Securities Australia Ltd	UBS Securities Australia Ltd	Sell	84,003		
UBS Securities Australia Ltd Sell 1,751 (186) UBS Securities Australia Ltd Sell 1,971 (180) UBS Securities Australia Ltd Buy 89,523 9,550 UBS AG London Branch Buy 89,217 9,550 UBS AG London Branch Buy 89,217 1,550 UBS AG London Branch Buy 89,417 1,550 UBS AG London Branch Buy 219,053 23,408 UBS AG London Branch Sell 219,073 1,635 UBS AG London Branch Sell 219,073 1,635 UBS AG London Branch Buy 1,874 1,073 UBS AG London Branch Buy 1,187 1,187 UBS Securities Australia Ltd Buy 1,187 1,159 1,25 UBS Securities Australia Ltd Buy 1,169 1,25 1,000 UBS Securities Australia Ltd Buy 1,636 1,159 1,25 UBS Securities Australia Ltd Buy 1,636 1,159 1,25 UBS	UBS Securities Australia Ltd	Sell	2,823		Ordinary
UBS Securities Australia Ltd Skil 1910 (203) UBS Securities Australia Ltd Buy 88,157 9,550 UBS AG London Branch Buy 88,412 9,550 UBS AG London Branch Buy 219,410 2,530 UBS AG London Branch Buy 219,410 2,3408 UBS AG London Branch Buy 219,410 1,324,08 UBS AG London Branch Buy 219,410 1,334,08 UBS AG London Branch Buy 219,410 1,334,08 UBS AG London Branch Buy 219,410 1,344 1,050,00 UBS AG London Branch Buy 219,410 1,344 1,050,00 UBS AG London Branch Buy 219,410 1,050,00 1,050,00 UBS Securities Australia Ltd Buy 1,150 1,150 1,150 UBS Securities Australia Ltd Buy 1,150 1,150 1,150 UBS Securities Australia Ltd Buy 1,150 1,150 1,150 UBS Securities Australia Ltd Buy 1,	UBS Securities Australia Ltd	Sell	1,751	(186)	Ordinary
UBS Securities Australia Ltd Sell 88 (4) UBS AG London Branch Sell Bly BS 277 9.550 UBS AG London Branch Sell Sell 2.19,053 9.550 UBS AG London Branch Bly Sell 2.19,053 2.3,408 UBS AG London Branch Bly Sell 2.19,053 2.3,408 UBS AG London Branch Bly Sell 2.19,053 2.3,408 UBS AG London Branch Bly Sell 2.19,010 2.13,300 2.3,408 UBS Securities Australia Ltd Bly Bly 1.0,414 1.0,200 9.550 UBS Securities Australia Ltd Bly 1.0,414 1.0,200 9.550 1.0,414 1.0,414 1.0,500 1.0,414 1.0,414 1.0,414 1.0,414 1.0,414 1.0,413 1.0,414 1.0,413 1.0,414 1.0,413 1.0,414 1.0,500 1.0,414 1.0,500 1.0,414 1.0,500 1.0,414 1.0,500 1.0,413 1.0,413 1.0,413 1.0,414 1.0,500 <t< td=""><td>UBS Securities Australia Ltd</td><td>Sell</td><td>1,910</td><td>(203)</td><td></td></t<>	UBS Securities Australia Ltd	Sell	1,910	(203)	
May Sequence Banch Bluy 89.55 9.550 UBS AG London Branch Bulk Bell 89.57 9.550 UBS AG London Branch Bulk Sell 219.033 23.408 UBS AG London Branch Bulk Sell 219.033 23.408 UBS AG London Branch Bulk Sell 219.010 23.408 UBS AG London Branch Sell 219.010 23.408 UBS Sequence Asstrated to the Branch Sell 219.010 23.408 UBS Securities Australia Ltd Buy 18.223 2.007 UBS Securities Australia Ltd Buy 18.23 2.007 UBS Securities Australia Ltd Buy 1.050 1.75 UBS Securities Australia Ltd Buy 1.050 1.75 UBS Securities Australia Ltd Buy 1.178 1.25 UBS Securities Australia Ltd Buy 1.169 1.25 UBS Securities Australia Ltd Buy 1.169 1.25 UBS Securities Australia Ltd Buy 1.169 1.169	JBS Securities Australia Ltd	Sell	38	(4)	
May Securities Australia Ltd Buy 1550 UBS Securities Australia Ltd Buy 219,038 23,409 UBS AG London Branch Sell 219,410 (23,409) UBS Securities Australia Ltd Buy 41,586 (23,409) UBS Securities Australia Ltd Buy 41,586 (6,113) UBS Securities Australia Ltd Buy 82,230 (6,113) UBS Securities Australia Ltd Buy 18,228 2,007 UBS Securities Australia Ltd Buy 88,258 (6,131) UBS Securities Australia Ltd Buy 88,258 (6,88) UBS Securities Australia Ltd Buy 1,178 1,25 UBS Securities Australia Ltd Buy 1,178 1,25 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy	JBS AG London Branch	Buy	89,557	9,550	
May 219,053 23,408 UBS AG London Branch Bluy 219,053 23,408 UBS Securities Australia Ltd Buy 1,194,10 (3,000,000) UBS Securities Australia Ltd Buy 1,134 (0,000,000) UBS Securities Australia Ltd Buy 1,0,414 (0,000,000) UBS Securities Australia Ltd Buy 1,0,53 (6,89) UBS Securities Australia Ltd Buy 1,0,23 (6,89) UBS Securities Australia Ltd Buy 1,0,35 (6,89) UBS Securities Australia Ltd Buy 1,0,35 (6,89) UBS Securities Australia Ltd Buy 1,0,35 (6,89) UBS Securities Australia Ltd Buy 1,0,59 1,25 UBS Securities Australia Ltd Buy 1,0,59 1,25 UBS Securities Australia Ltd	JBS AG London Branch	Sell	89,412	(9,550)	
UBS Securities Australia Ltd Seel 219,410 (22,408) UBS Securities Australia Ltd Boy 41,386 4,434 UBS Securities Australia Ltd Boy 41,386 4,434 UBS Securities Australia Ltd Boy 18,233 2,007 UBS Securities Australia Ltd Boy 18,233 1,550 UBS Securities Australia Ltd Boy 1,636 1,550 UBS Securities Australia Ltd Boy 1,636 1,55 UBS Securities Australia Ltd Boy 1,78 1,56 UBS Securities Australia Ltd Boy 1,78 1,78 UBS Securities Australia Ltd	JBS AG London Branch	Buy	219,053	23,408	
UBS Securities Australia Ltd Short returned NA (3,000,000) UBS Securities Australia Ltd Buy 41,586 4,534 UBS Securities Australia Ltd Buy 10,141 10,800 UBS Securities Australia Ltd Buy 18,223 20,77 UBS Securities Australia Ltd Buy 18,233 10,800 UBS Securities Australia Ltd Sell 64,288 16,889 UBS Securities Australia Ltd Buy 18,237 9,550 UBS Securities Australia Ltd Buy 1,639 1,73 UBS Securities Australia Ltd Buy 1,639 1,73 UBS Securities Australia Ltd Buy 1,639 1,73 UBS Securities Australia Ltd Buy 1,739 1,85 UBS Securities Australia Ltd Buy 1,739 1,85 UBS Securities Australia Ltd Buy 1,739 1,73 UBS Securities Australia Ltd Buy 1,739 1,73 UBS Securities Australia Ltd Buy 1,730 1,73 UBS Securities Australia Lt	JBS AG London Branch	l i	219,410	(23,408)	
UBS Securities Australia Ltd Buy 41,586 4,434 UBS Securities Australia Ltd Buy 1,134 1,0800 UBS Securities Australia Ltd Buy 101,414 1,0800 UBS Securities Australia Ltd Buy 18,223 2,007 UBS Securities Australia Ltd Buy 18,223 1,507 UBS Securities Australia Ltd Buy 18,233 1,527 UBS Securities Australia Ltd Buy 1,626 1,727 UBS Securities Australia Ltd Buy 1,628 1,727 UBS Securities Australia Ltd Buy 1,628 1,727 UBS Securities Australia Ltd Buy 1,178 1,25 UBS Securities Australia Ltd Buy 1,178 1,25 UBS Securities Australia Ltd Buy 1,178 1,28 UBS Securities Australia Ltd Buy 1,178 1,28 UBS Securities Australia Ltd Buy 1,178 1,24 UBS Securities Australia Ltd Buy 1,18 1,18 UBS Securities Australia Ltd	JBS Securities Australia Ltd	꽁	ΝΑ	(000'000'E)	
UBS Securities Australia Ltd Sell 57.330 (6.113) UBS Securities Australia Ltd Buy 10,414 10,000 UBS Securities Australia Ltd Buy 18,823 2,007 UBS Securities Australia Ltd Buy 18,823 2,007 UBS Securities Australia Ltd Sell 6,4298 (6,859) UBS Securities Australia Ltd Sell 6,433 (7,133) UBS Securities Australia Ltd Sell 4,4017 (4,713) UBS Securities Australia Ltd Buy 6,433 (8,527) UBS Securities Australia Ltd Buy 6,433 (8,527) UBS Securities Australia Ltd Buy 6,433 (8,527) UBS Securities Australia Ltd Buy 7,169 1,25 UBS Securities Australia Ltd Buy 7,169 1,25 UBS Securities Australia Ltd Buy 2,669 2,79 UBS Securities Australia Ltd Buy 2,669 2,79 UBS Securities Australia Ltd Buy 2,669 2,79 UBS Securities Australi	JBS Securities Australia Ltd	Buy	41,586	4,434	
UBS Securities Australia Ltd Buy 10,444 10,800 UBS Securities Australia Ltd Buy 18,823 10,277 UBS Securities Australia Ltd Buy 180,288 19,277 UBS Securities Australia Ltd Buy 86,256 (6,888) UBS Securities Australia Ltd Sell 44,017 (6,888) UBS Securities Australia Ltd Buy 89,557 9,550 UBS Securities Australia Ltd Buy 1,636 1,75 UBS Securities Australia Ltd Buy 1,636 1,75 UBS Securities Australia Ltd Buy 7,95 85 UBS Securities Australia Ltd Buy 7,19 7,95 UBS Securities Australia Ltd Buy 7,19 7,19 UBS Securities Australia Ltd Buy 7,19 7,19 UBS Securities Australia Ltd Buy <td>JBS Securities Australia Ltd</td> <td>Sell</td> <td>57,330</td> <td>(6,113)</td> <td></td>	JBS Securities Australia Ltd	Sell	57,330	(6,113)	
UBS Securities Australia Ltd Buy 18.823 2.077 UBS Securities Australia Ltd Sell 64,288 19,217 UBS Securities Australia Ltd Sell 64,288 (6,889) UBS Securities Australia Ltd Sell 219,410 9,550 UBS Securities Australia Ltd Sell 219,410 6,737 UBS Securities Australia Ltd Buy 6,631 645 UBS Securities Australia Ltd Buy 1,178 1,25 UBS Securities Australia Ltd Buy 1,178 1,25 UBS Securities Australia Ltd Buy 7,95 85 UBS Securities Australia Ltd Buy 7,95 85 UBS Securities Australia Ltd Buy 7,79 1,75 UBS Securities Australia Ltd Buy 7,79 1,85 UBS Securities Australia Ltd Buy 1,187 1,25 UBS Securities Australia Ltd Buy 1,187 1,25 UBS Securities Australia Ltd Buy 1,187 1,26 UBS Securities Australia Ltd Bu	JBS Securities Australia Ltd	Buy	101,414	10,800	Ordinary
UBS Securities Australia Ltd Buy 19,278 19,277 UBS Securities Australia Ltd Bell 64,288 16,288 UBS Securities Australia Ltd Buy 83,557 9,550 UBS Securities Australia Ltd Sell 34,939 1,737 UBS Securities Australia Ltd Buy 6,453 (688) UBS Securities Australia Ltd Buy 6,453 (688) UBS Securities Australia Ltd Buy 7,95 85 UBS Securities Australia Ltd Buy 7,98 96 UBS Securities Australia Ltd Buy 1,69 1,78 UBS Securities Australia Ltd Buy 1,	JBS Securities Australia Ltd	Buy	18,823	2,007	Ordinary
UBS Securities Australia Ltd Sell 64.296 (6.869) UBS Securities Australia Ltd Buy 86,527 9,520 UBS Securities Australia Ltd Sell 21,9470 (3,727) UBS Securities Australia Ltd Sell 44,017 (3,727) UBS Securities Australia Ltd Buy 1,633 (6,889) UBS Securities Australia Ltd Buy 1,639 1,25 UBS Securities Australia Ltd Buy 1,699 1,25 UBS Securities Australia Ltd Buy 7,95 85 UBS Securities Australia Ltd Buy 1,169 1,27 UBS Securities Australia Ltd Buy 1,187 1,24 UBS Securities Australia Ltd Buy 1,187 1,24 UBS Securities Australia Ltd Buy 1,187 1,24 UBS Securities Australia Ltd Buy	JBS Securities Australia Ltd	Buy	180,288	19,217	Ordinary
UBS Securities Australia Ltd Buy 98,557 9,550 UBS Securities Australia Ltd Sell 21,3410 (3,720) UBS Securities Australia Ltd Sell 44,017 (4,713) UBS Securities Australia Ltd Buy 6,635 (4,713) UBS Securities Australia Ltd Buy 1,636 1,25 UBS Securities Australia Ltd Buy 7,95 85 UBS Securities Australia Ltd Buy 7,95 85 UBS Securities Australia Ltd Buy 7,95 85 UBS Securities Australia Ltd Buy 7,79 85 UBS Securities Australia Ltd Buy 7,79 86 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,187 1,26 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,187 1,18 UBS Securities Australia Ltd Buy	JBS Securities Australia Ltd	Sell	64,298	(6,868)	Ordinary
UBS Securities Australia Ltd Sell 34,939 (3,727) UBS Securities Australia Ltd Sell 34,939 (3,727) UBS Securities Australia Ltd Buy (453 (688) UBS Securities Australia Ltd Buy (3,727) (47,713) UBS Securities Australia Ltd Buy (3,727) (688) UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 729 78 UBS Securities Australia Ltd Buy 729 72 UBS Securities Australia Ltd Buy 1,187	JBS Securities Australia Ltd	Buy	89,557	9,550	Ordinary
UBS Securities Australia Ltd Sell 44,017 (4,731) UBS Wealth Management Australia Ltd Sell 44,017 (4,731) (688) 175 (688) 105 175 (688) 105	JBS Securities Australia Ltd	Sell	219,410	(23,408)	
UBS Securities Australia Ltd Sell 44,017 (4,713) UBS Securities Australia Ltd Buy 1,635 175 UBS Securities Australia Ltd Buy 1,635 175 UBS Securities Australia Ltd Buy 1,169 1.25 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 7795 85 UBS Securities Australia Ltd Buy 7795 85 UBS Securities Australia Ltd Buy 4779 51 UBS Securities Australia Ltd Buy 1,169 125 UBS Securities Australia Ltd Buy 1,187 126 UBS Securities Australia Ltd Buy 4,198 96 UBS Securities Australia Ltd Buy 1,78 128 UBS Securities Australia Ltd Buy 1,178 126 UBS Securities Australia Ltd Buy 1,178 1,26 UBS Securities Australia Ltd Buy 1,178	JBS Securities Australia Ltd	lleS.	34,939	(3,727)	
UBS Securities Australia Ltd Sell 6,453 (688) UBS Securities Australia Ltd Buy 1,636 1,536 UBS Securities Australia Ltd Buy 1,169 1,55 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 7795 85 UBS Securities Australia Ltd Buy 7795 85 UBS Securities Australia Ltd Buy 7,169 1,56 UBS Securities Australia Ltd Buy 7,29 78 UBS Securities Australia Ltd Buy 7,29 78 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,172 1,27 UBS Securities Australia Ltd Buy 1,172	JBS Securities Australia Ltd	Sell	44,017	(4,713)	
UBS Securities Australia Ltd Buy 1,636 175 UBS Securities Australia Ltd Buy 6,031 125 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,180 1,178 UBS Securities Australia Ltd Buy 1,187 1,25 UBS Securities Australia Ltd Buy 1,178 1,24 UBS Securities Australia Ltd Buy 1,264	JBS Wealth Management Australia Ltd	Sell	6,453	(688)	Ordinary
UBS Securities Australia Ltd Buy 6,031 645 UBS Securities Australia Ltd Buy 1,178 125 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 2,609 279 UBS Securities Australia Ltd Buy 477 51 UBS Securities Australia Ltd Buy 477 51 UBS Securities Australia Ltd Buy 898 96 UBS Securities Australia Ltd Buy 1,169 125 UBS Securities Australia Ltd Buy 1,580 169 UBS Securities Australia Ltd Buy 1,780 126 UBS Securities Australia Ltd Buy 1,784 126 UBS Securities Australia Ltd Buy 1,784 126 UBS Securities Australia Ltd Buy 1,720 14 UBS Securities Australia Ltd Buy 1,724 124 UBS Securities Australia Ltd Buy 1,720 124 UBS Securities Australia Ltd Buy 1,720	JBS Securities Australia Ltd	Buy	1,636	175	Ordinary
UBS Securities Australia Ltd Buy 1,169 155 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 477 51 UBS Securities Australia Ltd Buy 898 96 UBS Securities Australia Ltd Buy 1,169 169 UBS Securities Australia Ltd Buy 4,198 96 UBS Securities Australia Ltd Buy 7,29 78 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,178 1,27 UBS Securities Australia Ltd Buy 1,178 1,27 UBS Securities Australia Ltd Buy 1,1720 1,84 UBS Securities Australia Ltd Buy 1,159	JBS Securities Australia Ltd	Buy	150'9	645	Ordinary
UBS Securities Australia Ltd Buy 1,178 126 UBS Securities Australia Ltd Buy 795 85 UBS Securities Australia Ltd Buy 2,609 279 UBS Securities Australia Ltd Buy 2,609 279 UBS Securities Australia Ltd Buy 898 96 UBS Securities Australia Ltd Buy 4,798 1,69 UBS Securities Australia Ltd Buy 4,798 449 UBS Securities Australia Ltd Buy 4,798 449 UBS Securities Australia Ltd Buy 4,798 449 UBS Securities Australia Ltd Buy 1,178 1,26 UBS Securities Australia Ltd Buy 6,545 700 UBS Securities Australia Ltd Buy 1,724 1,26 UBS Securities Australia Ltd Buy 1,724 1,09 UBS Securities Australia Ltd Buy 1,178 1,09 UBS Securities Australia Ltd Buy 1,179 1,24 UBS Securities Australia Ltd Buy 1,720<	JBS Securities Australia Ltd	Buy	1,169	125	Ordinary
UBS Securities Australia Ltd Buy 795 BS UBS Securities Australia Ltd Buy 2609 29 UBS Securities Australia Ltd Buy 477 51 UBS Securities Australia Ltd Buy 477 51 UBS Securities Australia Ltd Buy 898 96 UBS Securities Australia Ltd Buy 4,198 49 UBS Securities Australia Ltd Buy 4,198 49 UBS Securities Australia Ltd Buy 729 78 UBS Securities Australia Ltd Buy 1,187 126 UBS Securities Australia Ltd Buy 1,187 126 UBS Securities Australia Ltd Buy 1,178 1,26 UBS Securities Australia Ltd Buy 1,720 1,41 UBS Securities Australia Ltd Buy 1,720 1,42 UBS Securities Australia Ltd Buy 1,720 1,42 UBS Securities Australia Ltd Buy 1,720 1,44 UBS Securities Australia Ltd Buy 1,720 <	JBS Securities Australia Ltd	Buy	1.178	126	Ordinary
UBS Securities Australia Ltd Buy 795 B5 UBS Securities Australia Ltd Buy 2,609 279 UBS Securities Australia Ltd Buy 4,77 51 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 898 96 UBS Securities Australia Ltd Buy 4,798 449 UBS Securities Australia Ltd Buy 7,798 7,88 UBS Securities Australia Ltd Buy 7,178 4,48 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,187 1,02 UBS Securities Australia Ltd Buy 1,178 1,02 UBS Securities Australia Ltd Buy 1,178 1,02 UBS Securities Australia Ltd Buy 1,150 1,12 UBS Securities Australia Ltd Buy 1,150 1,12 UBS Securities Australia Ltd Buy 1,150 1,25 UBS Securities Australia Ltd Buy 1,15	JBS Securities Australia Ltd	Buy	795	85	Ordinary
UBS Securities Australia Ltd Buy 2,609 279 UBS Securities Australia Ltd Buy 477 51 UBS Securities Australia Ltd Buy 898 96 UBS Securities Australia Ltd Buy 1,698 96 UBS Securities Australia Ltd Buy 4,198 449 UBS Securities Australia Ltd Buy 4,198 449 UBS Securities Australia Ltd Buy 1,187 1,26 UBS Securities Australia Ltd Buy 1,178 1,26 UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 1,159 1,25 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,72	JBS Securities Australia Ltd	Buy	795	85	Ordinary
UBS Securities Australia Ltd Buy 477 51 UBS Securities Australia Ltd Buy 1,169 125 UBS Securities Australia Ltd Buy 898 96 UBS Securities Australia Ltd Buy 1,580 169 UBS Securities Australia Ltd Buy 7,79 78 UBS Securities Australia Ltd Buy 7,29 78 UBS Securities Australia Ltd Buy 5,24 56 UBS Securities Australia Ltd Buy 1,178 1,25 UBS Securities Australia Ltd Buy 1,734 1,26 UBS Securities Australia Ltd Buy 1,734 1,27 UBS Securities Australia Ltd Buy 1,734 1,72 UBS Securities Australia Ltd Buy 1,720 1,84 UBS Securities Australia Ltd Buy 1,720	JBS Securities Australia Ltd	Buy	2,609	279	Ordinary
UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 898 96 UBS Securities Australia Ltd Buy 1,580 169 UBS Securities Australia Ltd Buy 7,79 7,79 UBS Securities Australia Ltd Buy 7,79 7,79 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,78 1,05 UBS Securities Australia Ltd Buy 1,734 1,23 UBS Securities Australia Ltd Buy 1,524 1,02 UBS Securities Australia Ltd Buy 1,154 1,02 UBS Securities Australia Ltd Buy 1,156 1,02 UBS Securities Australia Ltd Buy 1,750 1,84 UBS Securities Australia Ltd Buy <td< td=""><td>JBS Securities Australia Ltd</td><td>Buv</td><td>477</td><td>51</td><td>Ordinary</td></td<>	JBS Securities Australia Ltd	Buv	477	51	Ordinary
UBS Securities Australia Ltd Buy 878 96 UBS Securities Australia Ltd Buy 878 96 UBS Securities Australia Ltd Buy 4,198 449 UBS Securities Australia Ltd Buy 7,29 78 UBS Securities Australia Ltd Buy 7,29 78 UBS Securities Australia Ltd Buy 1,178 126 UBS Securities Australia Ltd Buy 6,545 700 UBS Securities Australia Ltd Buy 1,178 1,24 UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720	IRS Securities Australia I td	8110	1160	125	Ordinan
UBS Securities Australia Ltd Buy 898 99 UBS Securities Australia Ltd Buy 1,580 169 UBS Securities Australia Ltd Buy 4,198 449 UBS Securities Australia Ltd Buy 729 78 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 5,24 56 UBS Securities Australia Ltd Buy 1,734 1,27 UBS Securities Australia Ltd Buy 1,734 1,22 UBS Securities Australia Ltd Buy 1,734 1,72 UBS Securities Australia Ltd Buy 1,724 1,029 UBS Securities Australia Ltd Buy 1,720 1,84 UBS Securities Australia Ltd Buy 1,7	IDA Copuration Automatica Lead		600	70	Ordinary
UBS Securities Australia Ltd Buy 1,580 199 UBS Securities Australia Ltd Buy 1,580 149 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 1,187 1,27 UBS Securities Australia Ltd Buy 5,24 1,6 UBS Securities Australia Ltd Buy 1,178 1,0 UBS Securities Australia Ltd Buy 1,154 1,234 1,2 UBS Securities Australia Ltd Buy 1,159 1,2 1 UBS Securities Australia Ltd Buy 1,159 1,2 1 UBS Securities Australia Ltd Buy 1,159 1,2 1 UBS Securities Australia Ltd Buy 1,150 1,2 1 UBS Securities Australia Ltd Buy 1,720 184	DDS SECULIFIES AUSTRALIA	an)	0 0	8	Ordinary
UBS Securities Australia Ltd Buy 1,580 169 UBS Securities Ltd Buy 4,198 499 UBS Securities Australia Ltd Buy 729 78 UBS Securities Australia Ltd Buy 1,187 127 UBS Securities Australia Ltd Buy 1,778 1,26 UBS Securities Australia Ltd Buy 6,545 700 UBS Securities Australia Ltd Buy 1,759 1,72 UBS Securities Australia Ltd Buy 1,759 1,75 UBS Securities Australia Ltd Buy 1,750 184 UBS Securities Australia Ltd Buy 1,750	JBS Securities Australia Ltd	BUJ	868	8	Ordinary
UBS Securities Australia Ltd Buy 4198 449 UBS Securities Australia Ltd Buy 7729 78 UBS Securities Australia Ltd Buy 1,187 127 UBS Securities Australia Ltd Buy 1,178 1,26 UBS Securities Australia Ltd Buy 6,545 700 UBS Securities Australia Ltd Buy 1,124 124 UBS Securities Australia Ltd Buy 1,524 120 UBS Securities Australia Ltd Buy 1,750 184 UBS Securities Australia Ltd Buy 1,750 <td>JBS Securities Australia Ltd</td> <td>Buy</td> <td>1,580</td> <td>169</td> <td>Ordinary</td>	JBS Securities Australia Ltd	Buy	1,580	169	Ordinary
UBS Securities Australia Ltd Buy 729 78 UBS Securities Australia Ltd Buy 1,187 127 UBS Securities Australia Ltd Buy 524 56 UBS Securities Australia Ltd Buy 1,178 126 UBS Securities Australia Ltd Buy 1,159 124 UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 1,564 2,0 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720	JBS Securities Australia Ltd	Buy	4,198	449	Ordinary
UBS Securities Australia Ltd Buy 1,187 127 UBS Securities Australia Ltd Buy 524 56 UBS Securities Australia Ltd Buy 1,778 56 UBS Securities Australia Ltd Buy 6,545 700 UBS Securities Australia Ltd Buy 1,534 132 UBS Securities Australia Ltd Buy 1,562 1,029 UBS Securities Australia Ltd Buy 1,750 1,144 UBS Securities Australia Ltd Buy 1,720 1,84 UBS Securities Australia Ltd Buy 1,553 1,34 UBS Securities Australia Ltd Buy <t< td=""><td>JBS Securities Australia Ltd</td><td>Buy</td><td>729</td><td>78</td><td>Ordinary</td></t<>	JBS Securities Australia Ltd	Buy	729	78	Ordinary
UBS Securities Australia Ltd Buy 524 56 UBS Securities Australia Ltd Buy 1,78 1,76 UBS Securities Australia Ltd Buy 6,545 700 UBS Securities Australia Ltd Buy 1,153 1,24 UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 1,964 2,10 UBS Securities Australia Ltd Buy 1,964 2,10 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,78 1,25 UBS Securities Australia Ltd Buy 1,78 1,25 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,16	JBS Securities Australia Ltd	Buv	1.187	127	Ordinary
UBS Securities Australia Ltd Buy 1,78 126 UBS Securities Australia Ltd Buy 1,784 700 UBS Securities Australia Ltd Buy 1,234 700 UBS Securities Australia Ltd Buy 1,789 1,23 UBS Securities Australia Ltd Buy 1,69 2 UBS Securities Australia Ltd Buy 1,70 184 UBS Securities Australia Ltd Buy 1,70 1,84 UBS Securities Australia Ltd Buy 1,70 1,84 UBS Securities Australia Ltd Buy 1,65 1,35 UBS Securities Australia Ltd Buy 1,66 1,25 UBS Securities Australia Ltd Buy 1,65	RS Securities Australia Ltd	Riv	524	35	Ordinary
UBS Securities Australia Ltd Buy 6.545 700 UBS Securities Australia Ltd Buy 6.545 700 UBS Securities Australia Ltd Buy 1,234 132 UBS Securities Australia Ltd Buy 9,621 1,029 UBS Securities Australia Ltd Buy 1,964 2.10 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,780 1,78 1,78 UBS Securities Australia Ltd Buy 1,78 1,78 1,78 UBS Securities Australia Ltd Buy 1,78 1,78 1,78 UBS Securities Australia Ltd Buy 1,78 1,79 1,79	IDC Cocurition Australia 14d	100	1,170	30,0	ordinary.
UBS Securities Australia Ltd Buy 1,545 700 UBS Securities Australia Ltd Buy 1,234 132 UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 1,621 1,029 UBS Securities Australia Ltd Buy 1,964 2,0 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,770 184 UBS Securities Australia Ltd Buy 1,780 1,26 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,69 1,24 UBS Securities Australia Ltd Buy 1	Jos Securities Australia Liu	- ina	9/1/1	97)	Ordinary
UBS Securities Australia Ltd Buy 1,234 132 UBS Securities Australia Ltd Buy 1,534 132 UBS Securities Australia Ltd Buy 9,621 1,029 UBS Securities Australia Ltd Buy 1,750 184 UBS Securities Australia Ltd Buy 1,750 184 UBS Securities Australia Ltd Buy 1,750 184 UBS Securities Australia Ltd Buy 1,770 184 UBS Securities Australia Ltd Buy 1,770 184 UBS Securities Australia Ltd Buy 1,780 1,26 UBS Securities Australia Ltd Buy 1,780 1,25 UBS Securities Australia Ltd Buy 1,780 1,25 UBS Securities Australia Ltd Buy 1,783 1,35 UBS Securities Australia Ltd Buy 1,553 1,34 UBS Securities Australia Ltd Buy 1,553 1,34 UBS Securities Australia Ltd Buy 1,653 1,34 UBS Securities Australia Ltd Buy <	JBS Securities Australia Ltd	Buy	6,545	700	Ordinary
UBS Securities Australia Ltd Buy 1,159 1,24 UBS Securities Australia Ltd Buy 96,21 1,029 UBS Securities Australia Ltd Buy 1,964 210 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,730 184 UBS Securities Australia Ltd Buy 1,730 184 UBS Securities Australia Ltd Buy 1,730 135 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,263 1,24 UBS Securities Australia Ltd Buy <t< td=""><td>JBS Securities Australia Ltd</td><td>Buy</td><td>1,234</td><td>132</td><td>Ordinary</td></t<>	JBS Securities Australia Ltd	Buy	1,234	132	Ordinary
UBS Securities Australia Ltd Buy 9,621 1,029 UBS Securities Australia Ltd Buy 19 2 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 134 UBS Securities Australia Ltd Buy 1,720 144 UBS Securities Australia Ltd Buy 1,778 126 UBS Securities Australia Ltd Buy 1,778 128 UBS Securities Australia Ltd Buy 1,169 135 UBS Securities Australia Ltd Buy 1,69 135 UBS Securities Australia Ltd Buy 1,63 134 UBS Securities Australia Ltd Buy 1,63 134 UBS Securities Australia Ltd Buy 1,63 134 UBS Securities Australia Ltd Buy 1,63 125 UBS Securities Australia Ltd Buy 1,63 125 UBS Securities Australia Ltd Buy 1,63 124 UBS Securities Australia Ltd Buy 1,63	JBS Securities Australia Ltd	Buy	1,159	124	Ordinary
UBS Securities Australia Ltd Buy 19 2 UBS Securities Australia Ltd Buy 1,964 210 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 2,179 2,799 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,603 1,25 UBS Securities Australia Ltd Buy 1,603 1,25 UBS Securities Australia Ltd Buy 1,603 1,25	JBS Securities Australia Ltd	Buv	9.671	1,029	Ordinary
UBS Securities Australia Ltd	IBS Societies Australia 14d	Blay	101	2011	, arding
UBS Securities Australia Ltd Buy 1,720 14 UBS Securities Australia Ltd Buy 1,720 125 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,78 1,78 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,169 1,25	IDC Constitute Australia 14d	Bin	1301	210	Ordinary
UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 2,179 2,799 233 UBS Securities Australia Ltd Buy 1,69 135 UBS Securities Australia Ltd Buy 1,539 134 UBS Securities Australia Ltd Buy 1,553 134 UBS Securities Australia Ltd Buy 1,169 125 UBS Securities Australia Ltd Buy 1,69 125 UBS Securities Australia Ltd Buy 1,603 125 UBS Securities Australia Ltd Buy 1,603 1,25	DO Securities Australia Ltd	ing d	406,1	017	Oldinary
UBS Securities Australia Ltd Buy 1,159 12 UBS Securities Australia Ltd Buy 1,778 126 UBS Securities Australia Ltd Buy 1,178 126 UBS Securities Australia Ltd Buy 2,179 233 UBS Securities Australia Ltd Buy 1,169 135 UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,169 125 1 UBS Securities Australia Ltd Buy 1,169 125 1 UBS Securities Australia Ltd Buy 1,003 125	DS Securities Australia Ltd	any	07/1	184	Ordinary
UBS Securities Australia Ltd Buy 1,720 184 UBS Securities Australia Ltd Buy 1,732 126 UBS Securities Australia Ltd Buy 2,179 233 UBS Securities Australia Ltd Buy 1,169 1,25 UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,169 125	JBS Securities Australia Ltd	Buy	691,1	125	Ordinary
UBS Securities Australia Ltd Buy 1,178 126 UBS Securities Australia Ltd Buy 2,179 233 UBS Securities Australia Ltd Buy 1,169 135 UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,169 135 UBS Securities Australia Ltd Buy 1,169 125 UBS Securities Australia Ltd Buy 1,169 125 UBS Securities Australia Ltd Buy 1,000 1,25	JBS Securities Australia Ltd	Buy	1,720	184	Ordinary
UBS Securities Australia Ltd Buy 2,179 233 UBS Securities Australia Ltd Buy 1,169 125 UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,169 125 I IRS Securities Australia Ltd Buy 1,053 125	JBS Securities Australia Ltd	8ny	1,178	126	Ordinary
UBS Securities Australia Ltd Buy 1,169 125 UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,169 1,25 1 IRS Securities Australia Ltd Buy 1,169 1,25	JBS Securities Australia Ltd	Buy	2,179	233	Ordinary
UBS Securities Australia Ltd Buy 1,253 134 UBS Securities Australia Ltd Buy 1,169 125 I IBS Securities Australia Ltd Buy 1,169 125 I IBS Securities Australia Ltd Buy 1,000 125	JBS Securities Australia Ltd	9uv	1,169	125	Ordinary
194 Oct 1950	IBC Counties Australia 14d	Bin	1 253	124	Continuo
1,169 1,25 1,25 1,25 1,25 1,25 1,25 1,25 1,25	DS Securities Australia Ltd	an)	567'1	134	Orginary
I IBS Secretion Australia 14	JBS Securities Australia Ltd	Buy	1,169	125	Ordinary
CBS Securities Australia Liu	JBS Securities Australia Ltd	Buy	1,992	213	Ordinary
1		UBS Securities Australia Ltd UBS Securities Australia Ltd	Self S	Sell Sell Sell Sell Sell Sell Sell Buy ll 2,823 Sell 1,910 Sock returned 1,586 Buy 1,178 Buy 1,109 Buy 1,100 Buy 1,1	

185 Securities Australia Ltd 04-Jun-14 UBS Securities Aus	Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	-	204	
	Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	727.6	700	Ordinary
	Buy Buy	27.077	2 899	Ordinary
	Buy Buy Buy Buy Buy Buy Buy Buy	5 128	549	Ordinacy
	Buy Buy Buy Buy Buy Buy Buy	9 545	1 022	Ordinan
	Buy Buy Buy Buy Buy Buy	11.199	1 199	Ordinary
	Buy Buy Buy Buy Buy	37.780	4.045	Ordinary
	Buy Buy Buy Buy	5.483	587	Ordinary
	Buy Buy Buy	17.251	1.847	Ordinary
	Buy Buy	1111	119	Ordinary
	Buy	1 149	123	Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(20)	V30	46	Ording of
1	Riv	1 513	16.3	Ordinary
	, , ,	5521	107	O'CILIER Y
	bud.	1012	5	Ordinary
	Buy	89.	18	Ordinary
	Buy	1,158	124	Ordinary
	Buy	467	20	Ordinary
	Buy	1,158	124	Ordinary
	Buy	6)706	986	Ordinary
	Buv	1.149	123	Ordinary
1	Buy	2560	9	Ordinary
	i i	1 140	25.	Ordinary Ordinary
	bug.	1,149	57	Cidinary
	any	CQ/,1	68	Ordinary
	Buy	961	21	Ordinary
	Buy	405	43	Ordinary
	Buy	2,802	300	Ordinary
	Buy	2,102	225	Ordinary
1	Su.	2.802	300	Ordinary
1	Six	10 694	1.145	Ordinary
	600	0000	103	Calibo
		50,00	5 .	, iii
	Buy	12,432	1,331	Ordinary
	Buy	33,465	3,583	Ordinary
1 1 1 1 1 1 1 1 1 1	Buy	1,541	165	Ordinary
1	Buy	22,481	2,407	Ordinary
1	Buv	15.803	1.692	Ordinary
	S ₁ ×	74 390	3 687	Ordinan
		101	1	
	buy	684	53	Ordinary
	Buy	663	7.1	Ordinary
	Buy	579	62	Ordinary
1 1 1 1 1 1 1 1 1 1	Buy	1,018	109	Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Rux	728	ΨV	Ordinary
	6	1007	9 5	dia di
	Buy	4,9/8	233	Ordinary
	Buy	1,803	193	Ordinary
	Buy	2,419	529	Ordinary
1	Buv	3.671	393	Ordinary
	Bris	111	700	Ordinary
	600	- CCC	F	Ordinary Ordinary
	ing	000,0	170	Orginary
	auy	9,335	1,000	Ordinary
	Buy	5,025	538	Ordinary
	Buy	7,229	774	Ordinary
	N B	5 557	707	Ordinan
	650	267.6	600	o di la la
	600	200,00	P S	2
	Buy	2,801	300	Orginary
	Buy	644	69	Ordinary
	Buy	7,089	759	Ordinary
	Buv	24.321	2.604	Ordinary
	Biv	7 182	769	Ordinary
	500	201,7	100	Cidinaly
	huy	3,801	40/	Ordinary
	Buy	50,128	2,36/	Ordinary
	Buy	15,327	1,641	Ordinary
П	Buy	3,755	402	Ordinary
	Biv	2 802	300	Ordinany
r	200	130	cot	in in it
Ť	, and	305	5	Ciciliary
7	Buy	4,222	457	Ordinary
04-Jun-14 UBS Securities Australia Ltd	Buy	1,588	170	Ordinary
Of the 44 TIPO Countries Australia 144		2001		

A A B B A A B B A A B B A A B B A A B B A A B B A A B B A B A B B A B A B B A B B A B B A B B A B B A B B B A B B B A B B B A B B B A B
A A B B B B B B B B B B B B B B B B B B
Buy
Buy
Buy

05-Jun-14	LUSS Securities Alistralia Ind				5
05-Jun-14	UBS Securities Australia Ltd	llas	124 522	(13.080)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	- S	9.425	(066)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	g	133	(14)	Ordinary
05. hm-17	I IDC Countition Australia 144	1100	CCC	(41)	500
05 Jun 14	I IDC Committee Australia 14d	100	250	(00)	Oldinary
100000	Coo Securities Australia Ltd	nel c	410,101	(60408)	Ordinary
02-Jun-14	UBS Securities Australia Ltd	Sell	1,304	(137)	Ordinar
05-Jun-14	UBS Securities Australia Ltd	Sell	2,056	(216)	Ordinary
US-JUN-14	UBS Securities Australia Ltd	ii ax	50,637	(5,319)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	48,761	(5,122)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	1,447	(152)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3.161	(332)	Ordinar
05-Jun-14	UBS Securities Australia Ltd	Sell	1 914	(201)	Ordinan
05- lun 14	Let Library A rotation O O O I	===	21.2.1	(102)	5
12-100-CO	ODS SECUTIONS AUSTRAINS LID	Sell	10,6/2	(1,121)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,637	(382)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	4,760	(200)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	162	(17)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	1,476	(155)	Ordinar
05-Jun-14	UBS Securities Australia Ltd	Sel	36 547	(3.839)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	ias	Uoc	(41)	Ordinary
05-lun-14	URS Securities Australia Ltd	9	1915	(056)	
10.00	Day Secondies Australia Live	261	101,0	(202)	Countries
05-Jun-14	UBS Securities Australia Ltd	Sell	5,664	(565)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,161	(332)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,161	(332)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	111,082	(11.656)	Ordinary
05-Jun-14	UBS Securities Australia 11d	lay	17 011	(1 785)	Ordinar
05- lim-14	I IDC Countries Australia (1)	3	VE2 22	(00/1)	100
ti-linc-co	Opp Securities Australia Ltd) sell	55,6/4	(5,842)	Ordinary
05-JUN-14	UBS Securities Australia Ltd	Sell	1,058	(111)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,164	(332)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	2,640	(277)	Ordinar
05-Jun-14	UBS Sequirities Australia I td	llaS	18717	(1 964)	Ordinan
05-lim-14	I IBS Sequities Australia 14d	= 0	039 6	(505)	
17	I IDC Cocurition Australia I to		0000	(000)	500
	ODD December Sharinging Fig.		9/	(o)	Gamary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,974	(417)	Ordinary
05-Jun-14	U8S Securities Australia Ltd	Sell	1,020	(107)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	10	Ξ	Ordinar
05-Jun-14	UBS Securities Australia Ltd	==Se=	553	(28)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	as a	10 588	(1111)	Ordinar
05. lin. 14	I IBS Societies Australia Ltd	le o	6 147	(545)	i i
05 11 14	Let Colored Access CO 201	100	1500 C	(0.1)	5 0
ti linc co	Obs securities Australia Liu) Seil	1,06,6	(014)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	5,718	(009)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,345	(321)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	5.928	(622)	Ordinar
05-1:n-14	LIBS Sequities Australia 11d	lay	5 718	(600)	cipio
1 1 1 10	1 100 C	150	01/20	(000)	5
US-JUN-14	UBS Securities Australia Ltd	Sell	3,116	(377)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Seli	3,164	(332)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	222,726	(13,371)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3.154	(331)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	lleS.	89.582	(9.400)	Ordinan
05-11m-14	I IRC Securities Australia 14d	Coll	2 164	(000)	C. C. C.
100	Lat of contrast A contrast of COL		101,0	(200)	500
	COO COUNTY OF THE PARTY OF THE	100	1,1	(150)	Oldinary
05-JUN-14	UBS Securities Australia Ltd	Sell	3,812	(400)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,154	(331)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,164	(332)	Ordinary
05-Jun-14	UBS Securities Australia 11d	lle?	37 373	(795.5)	Ordinan
05-km-14	I IBS Securities Australia 11d	100	3 183	(VEC)	Ordinan
	COO DESTRUCTION OF THE PROPERTY OF THE PROPERT	100	2,163	(324)	2000
03-Jun-14	Ubs securities Australia Ltd	Sell	3,154	(331)	Ordinary
US-Jun-14	UBS Securities Australia Ltd	Sell	3,164	(332)	Ordinar
05-Jun-14	UBS Securities Australia Ltd	Sell	105	(11)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	33,806	(3,551)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,808	(400)	Ordinan
05-Jun-14	UBS Securities Australia Ltd	Sel	1.276	(134)	Ordinary
05-lim-14	I (Of Congristing Australia 1+2	les.	027.7	1010/	2 2
15-JUN-14	UBS Securities Australia Ltd	Seil	99/'/	(816)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	9	4.760	(500)	Ordinan
				(nnc)	25
05-Jun-14	UBS Securities Australia Ltd	, in the second	27.018	(2838)	Ordinary

Buy 4,110	Γ	T	Τ	T	ŀ					_	7	1	Т	Т	T	Т	Т	Τ	Π	T	T	Т	Τ	П	1	Т	T	Π	Π	1	1	Τ	П	Т	Τ	Τ	П	Т	Т	T	T		-	Т	Т	Τ	1		П	Т	Т	Т	T	1	ľ	1	П
Buy 4,110 Buy 4,110 Buy 1,644 Buy 1,1479 Buy 4,110 Buy 4,110 Buy 4,110 Buy 4,110 Buy 4,110 Buy 4,110 Buy 7,539 Buy 7,539 Buy 1,7,469 Sell 1,7,469 Sell 1,7,469 Sell 1,7,469 Buy 7,531 Buy 7,531 Buy 7,531 Buy 7,6313 Buy 7,6313 Buy 7,340 Sell 1,7,469 Sell 1,7,469 Sell 1,7,469 Sell 1,7,464 Buy 7,130 Buy 7,130 Buy 7,231 Buy 7,240 Sell 1,449 Sell 1,440	-	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary		Ordinary
Buy Buy	3	176	1,229	526	965	440	382	300	593	80	(45)	(38)	(1,500)	4,593	(1835)	(503 /)	6623	(21,860)	3,030	1,904	(7,640)	1 835	(4,593)	4,000	(8,000)	676	(07.0)	(5,421)	(1,857)	(441)	(25,000)	(25,000)	(24,351)	(268)	(8,294)	(323)	(806)	(8,386)	(498)	(1,931)	(7,500)	(356)	(4)	(104)	(6.749)	(360)	(400)	(4,400)	(2,151)	(2)	(1,126)	(1,482)	(9/2)	(1,236)	(818)		(343)
nt and a second an	0,00	1,644	11.479	4,913	2,567	4,110	9,200	2,802	5,539	75	421	355	12,579	17.464	17 469	17,409	990'69	208,402	28,785	18,139	763.01	17 464	43,695	37,998	76,313	6,439	2.570	51,608	17,679	1 080	238,000	238,000	231,822	2,551	78.959	3,075	8,644	79,835	4,741	18.383	71,400	3,389	38	133	64.250	3,427	3,808	41,888	20,478	29	10,720	1 647	7,388	11,767	787,7		3,265
M1		Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Sell	Sell	3	Ang Ang	Soll	8 3	Buv	Sell	Buy	Buy	Sell	Buy	Sell	Buy	Sell	Buy Fotor Into Derivative	Sell	Sell	Sell	Sell	Sell	Sell	Self	Seli	N N	Sell	Sell	Sell	Sell	198	lles.	Sell	Sell	Jan 3	- See 1	Sell	Sell	Sell	Sell	Sell	Set:	Sell Sell	lles.	Sell	Sell		Sell
	T = 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S AG London Branch	S AG London Branch	S AG London Branch	S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	Securities Australia Ltd	S Securities Australia (1d	S Securities Australia Ltd	S Securities Australia Ltd			S Global Asset Management	UBS Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd				S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	5 Securities Australia Ltd	Securities Australia Ltd	S Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	5 Securities Australia Ltd		UBS Securities Australia Ltd

	801 1,773 95 505 507 18 107 1172 38,482 4,765 4,765 3,154 3,	(10) (10) (10) (10) (13) (13) (13) (13) (13) (13) (13) (13	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS S	95 1,773 95 505 124 5,718 1,172 1,172 1,172 4,055 4,055 4,055 4,055 4,055 4,055 4,055 3,154	(10) (10) (10) (10) (10) (10) (10) (10)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,773 5578 5578 124 50,708 1,172 38,482 4,765 3,154 3,15	(186) (100) (600) (600) (130) (120)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Lastralia Ltd	95 505 578 50404 1,122 38,482 38,482 38,482 3,154	(10) (53) (600) (13) (128) (4038) (4038) (331) (331) (344) (344) (344) (344) (344) (344) (344)	Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities UBS S	505 507 104 107 1172 38,482 38,482 3,154 3,	(62) (13) (12) (4,038) (500) (500) (331) (331) (344) (Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,718 124 50,404 1,172 4,765 4,765 4,765 4,765 3,154 3,1	(600) (5.289) (4.038) (600) (600) (631) (331) (331) (331) (331) (331) (331) (331) (331)	Ordinary Ordinary Ordinary
UBS Securries Australia Ltd UBS Securries Australia Ltd	20,004 1,172 1,172 1,175 1,154 3,154	(13) (123) (123) (4,038) (4,038) (500) (331) (331) (331) (331) (331) (331) (331) (331) (331) (331) (331)	Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Itd UBS Securities Itd UBS Securities Itd UBS Securities Itd UBS Securities Itd UBS Securities Itd	36,404 1,122 38,482 3,154 3,15	(5,289) (4,038) (500) (500) (331) (331) (331) (331) (331) (331) (331) (331)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	38.482 4,765 4,765 3,154 3,154 3,154 3,154 3,154 3,154 3,154 3,154 3,154 20,194 20,194 2,154 2,1	(4,038) (500) (500) (331) (331) (331) (331) (331) (331) (331) (331) (331)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	38,482 4,765 3,154 3,154 3,154 3,154 3,154 3,154 3,154 3,154 20,194 20,194 6,5471	(4,038) (500) (331) (331) (331) (320) (331) (320) (331) (2,119) (2,119)	
UBS Securities Australia Ltd UBS Securities Australia Ltd	3.154 3.154 3.154 3.154 3.154 3.154 3.154 2.20,194 3.154 2.20,194 3.154 2.20,194 3.154 2.20,194 3.154 2.20,194	(331) (331) (331) (328) (331) (320) (331) (331) (2,119) (2,119)	Crainary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3.154 3.154 3.154 3.154 3.154 3.278 3.20 3.050 3.050 3.154 2.0194 3.154 3.154	(331) (331) (331) (328) (331) (320) (331) (2,119) (2,119)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,154 3,154 3,134 3,134 3,134 3,134 3,154 2,019 3,154 2,0194 3,154 6,5471	(331) (3289) (331) (344) (320) (320) (321) (2,119) (2,119)	Ordinary Surgicial
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,154 3,154 3,154 3,154 3,154 20,194 20,194 3,154 6,547	(3.289) (3.289) (3.289) (3.31) (3.20) (3.20) (3.31) (2.119) (3.31)	Colling
UBS Securities Australia Ltd UBS Securities Ubstralia Ltd UBS Securities Ubstralia Ltd	31,344 3,154 3,154 3,020 3,020 20,194 3,154 3,154 3,154 3,154	(3,289) (3,289) (3,44) (3,20) (3,31) (3,31) (3,31)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,154 3,278 3,050 3,154 20,194 3,154 65,471	(331) (320) (320) (331) (331) (331)	Creiniary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,278 3,050 3,154 20,194 2,154 65,471	(344) (320) (331) (2,119) (331)	Colorary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,020 3,020 3,154 20,194 3,154 6,5471	(320) (321) (331) (331)	Crdinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,154 20,194 3,154 65,471	(331)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	20,104 20,194 3,154 65,471	(331)	Calling C
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,154 65,471	(331)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	65,471	(0.00)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2.154	- X C	Ordinary
UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securities Australia Ltd	1	(331)	Ordinary
UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securities Australia Ltd	4,984	(523)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	45,992	(4.826)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	257	(27)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,459	(363)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	57	(9)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	7,634	(801)	Ordinary
LBS Securities Australia Ltd LBS Securities Australia Ltd	305	(32)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,315.	(138)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	44,905	(4,712)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	25,150	(2,639)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,404	(267)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	4,860	(510)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	114	(12)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	20,899	(2,193)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	629	(99)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	161,1	(125)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	44,286	(4,647)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1,973	(202)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1,487	(156)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	18,965	(1,990)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	4,813	(505)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	133	(14)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	95	000	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	4,184	(439)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	248	(97)	Crainary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	15 730	(1 651)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	24 502	(7.571)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	3.755	(394)	Ordinary
UBS Securities Australia Ltd	13,885	(1,457)	Ordinary
UBS Securities Australia Ltd	76	(8)	Ordinary
	1,420	(149)	Ordinary
UBS Securities Australia Ltd	3,621	(380)	Ordinary
	4,365	(458)	Ordinary
UBS Securities Australia Ltd	1,191	(125)	Ordinary
5-Jun-14 UBS Securities Australia Ltd Sell	16,496	(1,731)	Ordinary
	01	Ξ	Ordinary
UBS Securities Australia Ltd	1,839	(193)	Ordinary
05-Jun-14 UBS Securities Australia Ltd Sell	9,540	(1,001)	Ordinary
UBS Securities Australia Ltd	52,301	(5,488)	Ordinary

								-	T	Τ	1	T	T	T	T	Τ	1			T	T				Τ	Τ	Π		T	Τ	Τ	П	T	Τ				T	T				Τ	Τ	Γ	Γ	Π	1	Т	T	Τ	Π	П	1	Т	_
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary
(421)	(8,264)	(1)	(508)	(53)	(985)	(331)	(297)	(53)	(40, E	(4)	(110)	(331)	(331)	(331)	(987)	(331)	(E)	(2,466)	(83)	(189)	(07)	(262)	(65)	(253)	(4)	(861)	(10)	(400)	(112)	(331)	(1,096)	(5)	(231)	(144)	(21)	(1,514)	(10)	(140)	(5,334)	(181)	(331)	(79)	(331)	(1,556)	(44)	(28)	(2)	(11)	(331)	(182)	(57)	(118)	(47)	(227)	(9)	(15)
4,008	78,673	01	4,836	276	9,349	3,151	2,827	505	/96'6	48	1,04/	3,151	3,151	101,0	968 6	3,151	29	23,476	790	796	895	5,664	295	2,409	3151	8,197	95	3,808	3 151	3,151	10,434	48	2,199	1,371	200	14,413	95	1,333	20,780	1,723	3,151	752	3.151	14,813	419	267	48	105	3,151	1.733	543	1,123	447	2,161	57	1.63.1
Sell	Sell	Setl	Sell	Sell	Sell	Sell	Sell	J. Kell	Jan Jan	Sell	Ser.	Sell	Jei Coll	- N	la S	3	Sell	Sell	Sell	Jake 19	¥ 34	Sell	Se¦	Zell Z	Ne le	lles.	Sell	Sell	Sell	Sell	Sell	Sell	<u> </u>	II-85	Sell		ja 3	- N	Sell	Sell	Sell	New York	les S	Sell	- Rel	Sell	====									
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	1005 Securities Australia Ltd	I IDC Coo critics Australia and	UBS Securities Abstralia Ltd	UBS Securities Australia 11d	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IDC Con witigs Australia Ltd																												
05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-30n-14	05-Jun-14	O5-JUN-14	05-Jun-14	05-July 14	05-lun-14	05-Jun-14		- 1	05-Jun-14	1	05-Jun-14																																				

		S S S S S S S S S S S S S S S S S S S	35,871 35,871 9,311 8,682 76	(510) (51) (3,764) (977)	Ordinary Ordinary Ordinary
		See See	35,871 9,311 8,682	(3,764)	Ordinary Ordinary
		See See	9,311 8,682 76	(977)	Ordinary
9			8,682	(116)	200
		See See	2,082	2	
					Ordinary
				0	Ordinary
			0000	(7/)	Orginary
		Se	46,068	(4,034)	Ordinary
		Sell	12,021	(0,52)	Oromary
		Sell	/06's	(410)	Ordinary
		Sell	29	(8)	Ordinary
		Sell Sell	496	(52)	Ordinary
		Sell	76	(8)	Ordinary
		Sell	3,812	(400)	Ordinary
			743	(78)	Ordinary
		Sell	3,183	(334)	Ordinary
		Seli	14,076	(1,477)	Ordinary
		Selt	305	(32)	Ordinary
		18	4,155	(436)	Ordinary
		les.	20,204	(2,120)	Ordinary
		B	1,125	(118)	Ordinary
		lleS.	43.285	(4.542)	Ordinary
		Sell	105	(11)	Ordinary
		les	238	(22)	Ordinary
1		Jac.	619	(65)	Ordinary
! !		Sell	2.256	(232)	Ordinary
		las	4 128	(430)	Ordinan
		9	1 584	(155)	Critical
		100	4 555	(101)	O O O
		100	43 222	(001)	Croiniary
	o Acstralia Etc	100	12,27	(00)	Old iai
$\Pi\Pi\Pi$	A Australia 4 al	130	045	(88)	Crainary
Ш	Australia Ltd	Jen Jen	050,5	(524)	Ordinary
П	A Australia Liu	lac o	3,302	(344)	Ordinary
7	s Australia Ltd	i i	9/6	(09)	Ordinary
	s Australia Ltd	le.	10,397	(1,083)	Ordinary
75-Jun-14 UBS Securities	UBS Securities Australia Ltd	Sell	3,554	(370)	Ordinary
T	s Australia Ltd	Sell	2,938	(306)	Ordinary
05-Jun-14 UBS Securities	JBS Securities Australia Ltd	Sell	6,211	(647)	Ordinary
	UBS Securities Australia Ltd	Sell	48	(2)	Ordinary
	s Australia Ltd	Sell	2,179	(227)	Ordinary
_	s Australia Ltd	Sell	2,688	(280)	Ordinary
	s Australia Ltd	Sell	1,402	(146)	Ordinary
5-Jun-14 UBS Securities	UBS Securities Australia Ltd	Sell	4310	(449)	Ordinary
Ī	JBS Securities Australia Ltd	ll d'S	2.246	(534)	Ording
Ī	Australia 11d	وا	1 238	(120)	Ordinary Croins
ľ	IBS Securities Australia 11d	Coll	2002	(201)	Gullary
1	Australia I to	100	1,450	(407)	Oldinary
1	Australia tid	liac.	1,459	(751)	Ordinary
05-Jun-14 Obs Securities	Library Australia Ho	liac li	817.7	(152)	Ordinary
1	A Australia Liu	130	0,770	(an/)	Ordinary
op-sum 4	Australia	liac l	7,650	(9/7)	Ordinary
1	Australia Ltd	les.	509	(63)	Ordinary
1	Australia Lid	Jet.	4,128	(430)	Ordinary
05-Juli-14 Obs Securities Australia Ltd	Australia Ltd	- N	355	(37)	Ordinary
-1	Australia Ltd	Sell	3,936	(410)	Ordinary
1	s Australia Ltd	Sell	864	(06)	Ordinary
_	s Australia Ltd	Sell	3,936	(410)	Ordinary
٦	s Australia Ltd	Seil	4,128	(430)	Ordinary
Jun-14 UBS Securities Australia Ltd	s Australia Ltd	Sell	1,565	(163)	Ordinary
1	s Australia Ltd	Sell	1,786	(186)	Ordinary
25-Jun-14 UBS Securities Australia Ltd	: Australia Ltd	Sell	4,128	(430)	Ordinary
	. Australia Ltd	les	2,150	(224)	Ordinary
05-Jun-14 UBS Securities Australia Ltd	: Australia Ltd	Sell	1,440	(150)	Ordinary
05-Jun-14 UBS Securities	: Australia Ltd	lleS	4,128	(430)	Ordinary
	Australia Ltd	lleS	8,064	(840)	Ordinary
ı	Australia : #d	Coll	523	(00)	in sign
US-Jun-14 IUBS Securities Australia Ltd		llex.	9/6	(09)	Ordinary

Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	nany	nary	nary	Jary	Jary	nany	nary	nary	nany	nary	nary	nary	nany	hary	hary	Jar.	harv	Jary	Jary	yary	Jary	Jary	nary Nary	Jary	Jar.	hany	yary	Jary	Jary	A Section	yan,	Jany	Jary	nary	any	ary	ary	any	lary	ary apy	ary any	any	iary	lary	lary	lary	ary eary	7	arv	ماد	T
(4) Ordi								(7) Ordi		i																Ordinary							1) Ordinary			3) Ordinary				L	1) Ordinary			0). Ordinary		1) Ordinary		S) Ordinary			Ordinary	ĺ	L		2) Ordinary		ĺ	L	L	7) Ordinary	
	(48	*	38)	(1,258)	(5,45	(400)	(4,211)	(3	(405)	(2,000)	E)	(576)	(32)	08)	(2)		(700)	51)	(130)	(35	(33	(33	6)	(26)	(33	(38)	(72	(273)	(40	(63)	(331)	(3,28	(331)	(331)	(1,500)	(27	(331)	(009)	(331)	(331)	(33)	(331)	(273)	(6,060)	(63)	(331)	(34	(7/8)	(4,101)	(757)	(115)	(106))	(12)	(29)	(800)	(50)	(157)	(2,427)	(0,000)
38	4,613	762	8,177	11,989	52,339	3,812	40,131	353	3,860	19,060	314	5,489	305	2,859	734	10	1/9/9	1,887	1,239	3,364	3,154	3,154	924	248	3 154	3.145	6,881	2,602	3,812	988	3,154	31,344	3 154	3,154	14,295	2,602	3,154	5,718	3,154	3 154	3,154	3,154	2,602	57,752	886	3,154	3,259	414	11 350	2 545	1.096	1,010	10	29	114	276	7 624	477	1,496	23,141	200
Self	Seli	Self	Self	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Seli	Self	Sell	Sell	Sell	Sell Sell	Sell	Sell	Sel.	Seil	Jes.	- S	الله الله	Sell	Sell	Set	Self	Sell	Jan Jan Jan Jan Jan Jan Jan Jan Jan Jan	la S	i d	Sell	Sell	llas	llas	Sell	Jest	les	Sell	Sell	Sell	Sell	Sel	Self	Sell	Sell	Jac J	197	Sell	Seil	Sell	Sell	Sell	Z Sell		Sell	Sell	Sell	
stralia Ltd	stralia Ltd	stralia Ltd	stralia Ltd	stralia Ltd	stralia Ltd	stralia Ltd	stralia Ltd	stralia Ltd	stralia Ltd	tralia Ltd	stralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	itralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	Iralia Ltd	tratta Ltd	tralia I td	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	trana Ltd	tralia +td	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia I td	tralia Itd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	Iralia Liu	rrafia I td	tralia Ltd	tralia Ltd	tralia Ltd	tralia Ltd	Iralia Ltd	ralia Ltd	ralia I td	ralia Ltd	ralia Ltd	ralia Ltd	
UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Aus	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Aus	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aust	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aug	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aust	UBS Securities Australia Ltd	UBS Securities Aust	UBS Securities Australia Ltd	UBS Securities Aust	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Aust	UBS Securities Australia Ltd	UBS Securities Australia Ltd	
- 1	- 1	- 1	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	1			05-Jun-14		1	05-Jun-14	ŀ	1	05-Jun-14	1	05-Jun-14	05-Jun-14	1	ı	05-Jun-14		- 1	- 1	05-JUN-34		05-Jun-14	-	l	05-Jun-14	-	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	05-Jun-14	1	05-Jun-14		05-Jun-14	05-Jun-14	05-Jun-14	05-lun-14	05-Jun-14	05-Jun-14	05-Jun-14	

One Systematical control of the control of	06-Jun-14	I IDC Commission A material to	ċ	000		
USS Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd	77-4	LIRC Sequition Australia 14d	N N	737 37	2007	Cleman
USS Cobels Assert Management Sell 1,979 770 USS Cobels Assert Management Autricle Ind Sell 6,139 777 USS Cobels Assert Management Autricle Ind Sell 6,139 6,73 USS Securities Autricle Ind Sell 6,23 6,73 USS Securities Autricle Ind Sell 7,72 6,93 USS Securities Autricle Ind Sell 1,22 1,73 USS Securities Autricle Ind Sell 1,23 1,63 USS Securities Autricle Ind Sell 1,23 1,63 USS Securities Autricle Ind Sell 1,12 1,12 USS Securities Autricle Ind Sell 2,59 1,59 USS Securities Autricle Ind Sell 2,59 1,59 USS Securities Autricle Ind Sell 1,52 1,44 USS Securities Autricle Ind Sell 2,59 1,50 USS Securities Autricle Ind Sell 2,59 1,50 USS Securities Autricle Ind Sell 1,50 1,50 USS Securities Autricle Ind <t< td=""><td>un-14</td><td>UBS Global Asset Management</td><td>Buv</td><td>49.857</td><td>5 1111</td><td>Ordinary</td></t<>	un-14	UBS Global Asset Management	Buv	49.857	5 1111	Ordinary
March Marc	un-14	UBS Global Asset Management	9	1 975	1000	diam'r.
USS Securities Australia Ltd Sell 6,599 (5,71) USS Securities Australia Ltd Boy 6,198 (6,71) USS Securities Australia Ltd Sell 2,590 (6,71) USS Securities Australia Ltd Sell 5,290 (6,71) USS Securities Australia Ltd Sell 1,223 (1,59) USS Securities Australia Ltd Sell 2,590 (1,10) USS Securities Australia Ltd Sell 1,100 (1,10) USS Securities Australia Ltd Sell 1,590 (1,10) USS Securities Australia Ltd Sell 1,500 (1,10) USS Securities	14	IRS Global Asset Management	B ₁₀	300 06	2 177	Guarany
March Marc	41-du	UBS Wealth Management Australia Ltd	وا	966 79	(4711)	Ordinary
UBS Securities Autorials Ltd Sell 2,550 2,720 1,055 2,047 2,527	un-14	UBS Wealth Management Australia Ltd	Buv	6.114	631	Ordinary
Loss Securities Australia Ltd Sell 26,560 (2,726) Loss Securities Australia Ltd Sell 5,29 (5,776) Loss Securities Australia Ltd Sell 5,29 (5,776) Loss Securities Australia Ltd Sell 656 (6,80) Loss Securities Australia Ltd Sell 1,323 (6,771) Loss Securities Australia Ltd Sell 1,323 (6,771) Loss Securities Australia Ltd Sell 1,1750 (1,159) Loss Securities Australia Ltd Sell 1,1750 (1,159) Loss Securities Australia Ltd Sell 1,1750 (1,199) Loss Securities Australia Ltd Sell 1,1720 (1,199) Loss Securities Australia Ltd Sell 1,1720 (1,190) Loss Securities Australia Ltd Sell 1,1720 (1,100)	un-14	UBS Securities Australia I td	Bit	20	7	ordina.
URS Securities Australia Ltd Bly 5,447 6.47 6.47 URS Securities Australia Ltd Sell 5,29 (5,4) URS Securities Australia Ltd Sell 1,323 (1,5) URS Securities Australia Ltd Sell 1,127 (1,15) URS Securities Australia Ltd Sell 1,172 (1,15) URS Securities Australia Ltd Sell 1,172 (1,15) URS Securities Australia Ltd Sell 1,172 (1,10) URS Securities Australia Ltd Sell 1,174 (1,17) URS Securities Australia Ltd Sell 1,174 (1,17) <	un-14	UBS Securities Australia Ltd	9	28.860	(307.0)	Colonia
Loss Securities Australia Ltd Sell 5.29 (5.7) Loss Securities Australia Ltd Sell 1,523 (5.7) Loss Securities Australia Ltd Sell 1,523 (15.5) Loss Securities Australia Ltd Sell 1,523 (15.6) Loss Securities Australia Ltd Sell 2,901 (15.6) Loss Securities Australia Ltd Sell 2,944 (17.73) Loss Securities Australia Ltd Sell 2,944 (17.73) Loss Securities Australia Ltd Sell 1,750 (1,199) Loss Securities Australia Ltd Sell 1,750 (1,199) Loss Securities Australia Ltd Sell 1,525 (1,006) Loss Securities Australia Ltd Sell 1,520 (1,006) Loss Securities Australia Ltd Sell 1,520 (1,006) Loss Securities Australia Ltd Sell 5,78 (1,072) Loss Securities Australia Ltd Sell 1,025 (1,006) Loss Securities Australia Ltd Sell 1,025 (1,046)	un-14	UBS Securities Australia Ltd	Res.	5 447	557	, decipor
URS Securities Australia Ltd Sell 1,523 (57) URS Securities Australia Ltd Sell 1,122 (15) URS Securities Australia Ltd Sell 1,127 (15) URS Securities Australia Ltd Sell 1,127 (21) URS Securities Australia Ltd Sell 1,129 (1,29) URS Securities Australia Ltd Sell 1,179 (1,70) URS Securities Australia Ltd Sell 1,174 (1,174) URS Securities Australia Ltd Sell 1,174 (1,174) URS Securities Australia Ltd Sell 1,174 (1,174) URS Securities Aus		I IBC Congretion Australia 14d	653	001	45	O'CILLIAN)
UBS Securities Australia Ltd Sell 1,323 (135) UBS Securities Australia Ltd Sell 1,323 (135) UBS Securities Australia Ltd Sell 2,155 (156) UBS Securities Australia Ltd Sell 2,901 (256) UBS Securities Australia Ltd Sell 2,901 (726) UBS Securities Australia Ltd Sell 1,1750 (1,729) UBS Securities Australia Ltd Sell 1,1760 (1,729) UBS Securities Australia Ltd Sell 1,1760 (1,729) UBS Securities Australia Ltd Sell 1,1760 (1,729) UBS Securities Australia Ltd Sell 1,2,256 (1,727) UBS Securities Australia Ltd Sell 1,2,256 (1,727) UBS Securities Australia Ltd Sell 1,2,256 (1,972) UBS Securities Australia Ltd Sell 1,2,256 (1,972) UBS Securities Australia Ltd Sell 1,1,44 (1,1,272) UBS Securities Australia Ltd Sell 1,072 (1,009)	17	I IBC Societies Australia I to	le de	973	(54)	Ordinary
UBS Securities Australia Ltd Sell 1,127 (159) UBS Securities Australia Ltd Sell 5,911 (526) UBS Securities Australia Ltd Sell 5,911 (526) UBS Securities Australia Ltd Sell 1,177 (115) UBS Securities Australia Ltd Sell 1,176 (526) UBS Securities Australia Ltd Sell 1,272 (446) UBS Securities Australia Ltd Sell 1,282 (1,072) UBS Securities Australia Ltd Sell 1,382 (1,049) UBS Securities Australia Ltd Sell 1,522 (60) UBS Securities Australia Ltd Sell 1,522 (63) UBS Securities Australia Ltd Sell 1,522 (63) UBS Securities Aust	1	I IRC Con white Australia 1-4	3	1 222	(70)	Colling
UBS Securities Australia Ltd Sell 1,170 (115) UBS Securities Australia Ltd Sell 2,155 (256) UBS Securities Australia Ltd Sell 2,971 (246) UBS Securities Australia Ltd Sell 2,971 (446) UBS Securities Australia Ltd Sell 2,964 (1,729) UBS Securities Australia Ltd Sell 1,760 (1,139) UBS Securities Australia Ltd Sell 1,760 (1,179) UBS Securities Australia Ltd Sell 1,285 (1,672) UBS Securities Australia Ltd Sell 1,286 (1,572) UBS Securities Australia Ltd Sell 1,286 (1,572) UBS Securities Australia Ltd Sell 1,286 (1,672) UBS Securities Australia Ltd Sell 1,046 (1,692) UBS Se	1 2	Life Commission Australia Ltd	Jell Jell	525,1	(551)	Ordinary
UBS Securities Australia Ltd Sell 5,177 (175) UBS Securities Australia Ltd Sell 2,901 (296) UBS Securities Australia Ltd Sell 4,377 (440) UBS Securities Australia Ltd Sell 1,759 (1,799) UBS Securities Australia Ltd Sell 1,760 (1,719) UBS Securities Australia Ltd Sell 1,286 (1,000) UBS Securities Australia Ltd Sell 1,520 (1,000) UBS	:	Obs Securities Australia Ltd	- X	999	(99)	Ordinary
UBS Securities Australia Ltd Sell 2,315 (256) UBS Securities Australia Ltd Sell 2,301 (256) UBS Securities Australia Ltd Sell 2,600 (1,730) UBS Securities Australia Ltd Sell 1,750 (1,179) UBS Securities Australia Ltd Sell 1,625 (1,006) UBS Securities Australia Ltd Sell 1,625 (1,006) UBS Securities Australia Ltd Sell 1,286 (1,007) UBS Securities Australia Ltd Sell 1,200 (1,137) UBS Securities Australia Ltd Sell 1,500 (1,049) UBS Securities Australia Ltd Sell 1,045 (1,049) UBS Securities Australia Ltd Sell 1,045 (1,049) UBS Securities Australia Ltd Sell 1,045 (1,049) UB	1	UBS Securities Australia Ltd	Jell	/71'1	(115)	Ordinary
UBS Securities Australia Ltd Sell 4,371 (446) UBS Securities Australia Ltd Sell 4,371 (448) UBS Securities Australia Ltd Sell 1,759 (1,000) UBS Securities Australia Ltd Sell 1,759 (1,000) UBS Securities Australia Ltd Sell 1,759 (1,000) UBS Securities Australia Ltd Sell 1,382 (1,000) UBS Securities Australia Ltd Sell 1,382 (1,000) UBS Securities Australia Ltd Sell 2,286 (1,000) UBS Securities Australia Ltd Sell 1,382 (1,000) UBS Securities Australia Ltd Sell 1,360 (1,000) UBS Securities Australia Ltd Sell 1,300 (1,000) UBS Securities Australia Ltd Sell 1,300 (1,000) UB	41-14	UBS Securities Australia Ltd	Sell	5,155	(526)	Ordinary
UBS Securities Australia Ltd Sell 2,460 (251) UBS Securities Australia Ltd Sell 2,460 (1,791) UBS Securities Australia Ltd Sell 1,750 (1,729) UBS Securities Australia Ltd Sell 1,652 (1,006) UBS Securities Australia Ltd Sell 1,750 (1,41) UBS Securities Australia Ltd Sell 2,185 (1,006) UBS Securities Australia Ltd Sell 1,2,256 (2,271) UBS Securities Australia Ltd Sell 1,2,256 (1,007) UBS Securities Australia Ltd Sell 1,0,270 (1,049)	un-14	UBS Securities Australia Ltd	Seli	2,901	(362)	Ordinary
UBS Securities Australia Ltd Sell 1,2460 (1,519) UBS Securities Australia Ltd Sell 1,729 (1,009) UBS Securities Australia Ltd Sell 1,524 (1,009) UBS Securities Australia Ltd Sell 1,525 (1,0074) UBS Securities Australia Ltd Sell 2,255-4 (2,337) UBS Securities Australia Ltd Sell 2,255-4 (2,337) UBS Securities Australia Ltd Sell 2,256-4 (2,337) UBS Securities Australia Ltd Sell 1,326 (1,572) UBS Securities Australia Ltd Sell 1,560 (1,694) UBS Securities Australia Ltd Sell 5,983 (3,973) UBS Securities Australia Ltd Sell 5,983 (3,973) UBS Securities Australia Ltd Sell 5,983 (3,973) UBS Securities Australia Ltd Sell 3,945 (1,049) UBS Securities Australia Ltd Sell 3,945 (1,049) UBS Securities Australia Ltd Sell 3,528 (5,950)	un-14	UBS Securities Australia Ltd	Sell	4,371	(446)	Ordinary
UBS Securities Australia Ltd Sell 11,550 (1,199) UBS Securities Australia Ltd Sell 1,750 (1,729) UBS Securities Australia Ltd Sell 1,624 (1,729) UBS Securities Australia Ltd Sell 2,185 (1,729) UBS Securities Australia Ltd Sell 2,185 (1,729) UBS Securities Australia Ltd Sell 2,255 (2,271) UBS Securities Australia Ltd Sell 1,526 (1,729) UBS Securities Australia Ltd Sell 1,526 (1,694) UBS Securities Australia Ltd Sell 1,582 (1,694) <	un-14	UBS Securities Australia Ltd	Sell	2,460	(251)	Ordinary
UBS Securities Australia Ltd Sell 1 (5,944 (1,029) UBS Securities Australia Ltd Sell 1 (5,944 (1,079) UBS Securities Australia Ltd Sell 1 (1,072) (1,072) UBS Securities Australia Ltd Sell 2,186 (2,233) UBS Securities Australia Ltd Sell 2,285 (2,237) UBS Securities Australia Ltd Sell 1,325 (1,072) UBS Securities Australia Ltd Sell 1,325 (1,072) UBS Securities Australia Ltd Sell 1,525 (1,072) UBS Securities Australia Ltd Sell 1,560 (6,09) UBS Securities Australia Ltd Sell 1,560 (6,59) UBS Securities Australia Ltd Sell 1,1,144 (1,134) UBS Securities Australia Ltd Sell 1,502 (5,590)	un-14	UBS Securities Australia Ltd	Sell	11,750	(1,199)	Ordinary
UBS Securities Australia Ltd Sell 16,934 (1,729) UBS Securities Australia Ltd Sell 2,185 (1,724) UBS Securities Australia Ltd Sell 2,185 (1,724) UBS Securities Australia Ltd Sell 2,235 (1,737) UBS Securities Australia Ltd Sell 2,236 (1,737) UBS Securities Australia Ltd Sell 1,382 (1,737) UBS Securities Australia Ltd Sell 1,382 (1,737) UBS Securities Australia Ltd Sell 1,362 (1,693) UBS Securities Australia Ltd Sell 1,582 (1,693) UBS Securities Australia Ltd Sell 1,981 (1,693) UBS Securities Australia Ltd Sell 1,981 (1,693) <	36-Jun-14	UBS Securities Australia Ltd	Sell	658'6	(1,006)	Ordinary
UBS Securities Australia Ltd Sell 1,052 (1,074) UBS Securities Australia Ltd Sell 2,188 (2,233) UBS Securities Australia Ltd Sell 2,2,854 (2,237) UBS Securities Australia Ltd Sell 1,382 (1,972) UBS Securities Australia Ltd Sell 1,282 (1,972) UBS Securities Australia Ltd Sell 1,282 (1,049) UBS Securities Australia Ltd Sell 1,282 (1,049) UBS Securities Australia Ltd Sell 1,286 (1,049) UBS Securities Australia Ltd Sell 1,286 (1,049) UBS Securities Australia Ltd Sell 1,286 (1,049) UBS Securities Australia Ltd Sell 1,347 (1,049) UBS Securities Australia Ltd Sell 1,347 (1,049) UBS Securities Australia Ltd Sell 1,348 (1,59) UBS Securities Australia Ltd Sell 1,348 (1,59) UBS Securities Australia Ltd Sell 1,348 (1,59) <td< td=""><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>16,944</td><td>(1,729)</td><td>Ordinary</td></td<>	un-14	UBS Securities Australia Ltd	Sell	16,944	(1,729)	Ordinary
UBS Securities Australia Ltd Sell 2,185 (2,23) UBS Securities Australia Ltd Sell 2,186 (2,23) UBS Securities Australia Ltd Sell 2,2,26 (2,27) UBS Securities Australia Ltd Sell 1,382 (1,41) UBS Securities Australia Ltd Sell 1,932 (1,697) UBS Securities Australia Ltd Sell 1,526 (1,694) UBS Securities Australia Ltd Sell 1,025 (1,049) UBS Securities Australia Ltd Sell 3,045 (311) UBS Securities Australia Ltd Sell 1,025 (1,049) UBS Securities Australia Ltd Sell 1,027 (1,049) UBS Securities Australia Ltd Sell 1,027 (1,049) UBS Securities Australia Ltd Sell 1,1,1,14 (1,1,1,14) UBS Securities Australia Ltd Sell 1,1,1,14 (1,1,2,1) UBS Securities Australia Ltd Sell 1,2,1,1,14 (1,1,2,1) UBS Securities Australia Ltd Sell 2,1,2,1 (1,2,2)	un-14	UBS Securities Australia Ltd	Sell	10,525	(1,074)	Ordinary
UBS Securities Australia Ltd Sell 2,185 (2,23) UBS Securities Australia Ltd Sell 1,326 (1,41) UBS Securities Australia Ltd Sell 1,326 (1,271) UBS Securities Australia Ltd Sell 5,933 (1,572) UBS Securities Australia Ltd Sell 5,933 (5,204) UBS Securities Australia Ltd Sell 5,933 (6,204) UBS Securities Australia Ltd Sell 5,933 (6,204) UBS Securities Australia Ltd Sell 1,025 (1,049) UBS Securities Australia Ltd Sell 1,027 (1,049) UBS Securities Australia Ltd Sell 1,027 (1,049) UBS Securities Australia Ltd Sell 1,027 (1,049) UBS Securities Australia Ltd Sell 1,144 (1,1347) UBS Securities Australia Ltd Sell 1,303 (1,049) UBS Securities Australia Ltd Sell 1,303 (1,049) UBS Securities Australia Ltd Sell 1,303 (1,049) <td< td=""><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Seil</td><td>431</td><td>(44)</td><td>Ordinary</td></td<>	un-14	UBS Securities Australia Ltd	Seil	431	(44)	Ordinary
UBS Securities Australia Ltd Sell 22,854 (2,332) UBS Securities Australia Ltd Sell 22,856 (2,332) UBS Securities Australia Ltd Sell 22,856 (2,044) UBS Securities Australia Ltd Sell 19,326 (1,972) UBS Securities Australia Ltd Sell 10,256 (1,694) UBS Securities Australia Ltd Sell 10,256 (1,694) UBS Securities Australia Ltd Sell 10,270 (1,049) UBS Securities Australia Ltd Sell 10,270 (1,049) UBS Securities Australia Ltd Sell 1,366 (1,037) UBS Securities Australia Ltd Sell 1,374 (1,346) UBS Securities Australia Ltd Sell 1,311 (1,337) UBS Securities Australia Ltd Sell 2,348 (591) UBS Securities Australia Ltd Sell 2,348 (675) UBS Securities Australia Ltd Sell 2,348 (675) UBS Securities Australia Ltd Sell 2,348 (675) <t< td=""><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Seil</td><td>7 185</td><td>(223)</td><td>Ordinary</td></t<>	un-14	UBS Securities Australia Ltd	Seil	7 185	(223)	Ordinary
UBS Securities Australia Ltd Seil 1,382 (177) UBS Securities Australia Ltd Seil 1,3256 (1,271) UBS Securities Australia Ltd Seil 19,326 (1,921) UBS Securities Australia Ltd Seil 10,973 (609) UBS Securities Australia Ltd Seil 10,270 (1,046) UBS Securities Australia Ltd Seil 10,270 (1,046) UBS Securities Australia Ltd Seil 10,270 (1,046) UBS Securities Australia Ltd Seil 10,270 (1,049) UBS Securities Australia Ltd Seil 1,566 (1,046) UBS Securities Australia Ltd Seil 1,566 (1,046) UBS Securities Australia Ltd Seil 1,11,144 (1,1347) UBS Securities Australia Ltd Seil 2,519 (257) UBS Securities Australia Ltd Seil 2,519 (277) UBS Securities Australia Ltd Seil Seil (678) UBS Securities Australia Ltd Seil Seil (678) U	un-14	UBS Securities Australia Ltd	Sell	22.854	(2 332)	Ordina
UBS Securities Australia Ltd Sell 22,256 (2,277) UBS Securities Australia Ltd Sell 19,326 (1,972) UBS Securities Australia Ltd Sell 60,973 (6,09) UBS Securities Australia Ltd Sell 5,069 (1,049) UBS Securities Australia Ltd Sell 10,270 (1,049) UBS Securities Australia Ltd Sell 1,374 (1,049) UBS Securities Australia Ltd Sell 1,374 (1,049) UBS Securities Australia Ltd Sell 1,911 (1,57) UBS Securities Australia Ltd Sell 1,911 (1,57) UBS Securities Australia Ltd Sell 5,912 (2,57) UBS Securities Australia Ltd Sell 5,912 (2,57) UBS Securities Australia Ltd Sell 5,912 (2,57) <	un-14	UBS Securities Australia 11d	اق	1 382	(141)	Ordinary
USS Securities Australia Ltd Sell 19,326 (1,972) UBS Securities Australia Ltd Sell 5,968 (6,044) UBS Securities Australia Ltd Sell 5,968 (6,044) UBS Securities Australia Ltd Sell 10,251 (1,049) UBS Securities Australia Ltd Sell 10,257 (1,049) UBS Securities Australia Ltd Sell 10,270 (1,049) UBS Securities Australia Ltd Sell 10,270 (1,049) UBS Securities Australia Ltd Sell 11,144 (1,137) UBS Securities Australia Ltd Sell 11,144 (1,137) UBS Securities Australia Ltd Sell 29,282 (59,942) UBS Securities Australia Ltd Sell 2,548 (675) UBS Securities Australia Ltd Sell 2,548 (675) UBS Securities Australia Ltd Sell 2,548 (675) UBS Securities Australia Ltd Sell 2,549 (727) UBS Securities Australia Ltd Sell 2,549 (727) <td< td=""><td>14</td><td>I BS Securities Australia I td</td><td>les</td><td>320 00</td><td>(175,57)</td><td>Ordinary</td></td<>	14	I BS Securities Australia I td	les	320 00	(175,57)	Ordinary
UBS Securities Australia Ltd Set 1,320 (1,57) UBS Securities Australia Ltd Sell 5,973 (5,204) UBS Securities Australia Ltd Sell 5,973 (5,204) UBS Securities Australia Ltd Sell 10,251 (1,046) UBS Securities Australia Ltd Sell 10,257 (1,046) UBS Securities Australia Ltd Sell 1,566 (160) UBS Securities Australia Ltd Sell 1,566 (160) UBS Securities Australia Ltd Sell 1,566 (160) UBS Securities Australia Ltd Sell 1,1,44 (11,347) UBS Securities Australia Ltd Sell 1,911 (1,347) UBS Securities Australia Ltd Sell 1,911 (1,505) UBS Securities Australia Ltd Sell 1,911 (1,507) UBS Securities Australia Ltd Sell 5,828 (391) UBS Securities Australia Ltd Sell 5,828 (391) UBS Securities Australia Ltd Sell 5,828 (391) UBS Secu		to Consider Australia Ltd	lion of	062,22	(1,2,2)	Ordinary
UBS Securities Australia Ltd Sell 50,973 (5,204) UBS Securities Australia Ltd Sell 16,507 (1,644) UBS Securities Australia Ltd Sell 10,251 (1,046) UBS Securities Australia Ltd Sell 10,257 (1,046) UBS Securities Australia Ltd Sell 1,566 (1,609) UBS Securities Australia Ltd Sell 1,1,144 (1,609) UBS Securities Australia Ltd Sell 1,1,144 (1,609) UBS Securities Australia Ltd Sell 2,338 (595) UBS Securities Australia Ltd Sell 1,911 (1,675) UBS Securities Australia Ltd Sell 2,948 (811) UBS Securities Australia Ltd Sell 1,911 (1,675) UBS Securities Australia Ltd Sell 2,538 (895) UBS Securities Australia Ltd Sell 2,516 (1,675) UBS Securities Australia Ltd Sell 2,516 (1,675) UBS Securities Australia Ltd Sell 2,516 (1,675)		1 DC Committee Australia Ltd	lio o	925,61	(7/6'1)	Ordinary
UBS Securities Australia Ltd Sell 16,004 UBS Securities Australia Ltd Sell 16,001 UBS Securities Australia Ltd Sell 10,251 (1,049) UBS Securities Australia Ltd Sell 10,270 (1,049) UBS Securities Australia Ltd Sell 1,566 (1609) UBS Securities Australia Ltd Sell 5,828 (599) UBS Securities Australia Ltd Sell 11,144 (1,1347) UBS Securities Australia Ltd Sell 11,144 (1,1347) UBS Securities Australia Ltd Sell 2,93,282 (599) UBS Securities Australia Ltd Sell 1,911 (1,155) UBS Securities Australia Ltd Sell 5,218 (59,21) UBS Securities Australia Ltd Sell 5,219 (52,7) UBS Securities Australia Ltd		Opp Securities Australia Ltd	i ac	50	(c)	Ordinary
UBS Securities Australia Ltd Sell 16,601 (1,649) UBS Securities Australia Ltd Sell 10,251 (1,049) UBS Securities Australia Ltd Sell 10,257 (1,049) UBS Securities Australia Ltd Sell 1,566 (1609) UBS Securities Australia Ltd Sell 1,566 (1609) UBS Securities Australia Ltd Sell 1,1,44 (1,1,347) UBS Securities Australia Ltd Sell 1,1,44 (1,1,347) UBS Securities Australia Ltd Sell 2,932,82 (855) UBS Securities Australia Ltd Sell 7,948 (817) UBS Securities Australia Ltd Sell 7,948 (817) UBS Securities Australia Ltd Sell 2,519 (875) UBS Securities Australia Ltd Sell 2,561 (875) UBS Securities Australia Ltd Sell 2,566 (150) UBS Securities Australia Ltd Sell 2,166 (150) UBS Securities Australia Ltd Sell 2,166 (150) UBS	* :	ODS Securities Australia Ltd	Seil	5/6/04	(5,204)	Ordinary
UBS Securities Australia Ltd Sell 1,568 (609) UBS Securities Australia Ltd Sell 1,025 (1,046) UBS Securities Australia Ltd Sell 1,027 (1,049) UBS Securities Australia Ltd Sell 1,566 (1,049) UBS Securities Australia Ltd Sell 5,78 (1,049) UBS Securities Australia Ltd Sell 1,11,144 (1,137) UBS Securities Australia Ltd Sell 2,942 (1,87) UBS Securities Australia Ltd Sell 2,942 (1,87) UBS Securities Australia Ltd Sell 2,942 (1,87) UBS Securities Australia Ltd Sell 2,546 (6,18) UBS Securities Australia Ltd Sell 2,546 (6,18) UBS Securities Australia Ltd Sell 2,546 (1,27) UBS Securities Australia Ltd Sell 2,546 (1,27) UBS Securities Australia Ltd Sell 2,546 (1,27) UBS Securities Australia Ltd Sell 2,340 (1,29) UBS Sec	-14 -14	UBS Securities Australia Ltd	Sell	16,601	(1,694)	Ordinary
UBS Securities Australia Ltd Sell 10,251 (1,046) UBS Securities Australia Ltd Sell 3,045 (1,049) UBS Securities Australia Ltd Sell 1,566 (160) UBS Securities Australia Ltd Sell 5,78 (595) UBS Securities Australia Ltd Sell 1,144 (1,1,347) UBS Securities Australia Ltd Sell 2,948 (811) UBS Securities Australia Ltd Sell 7,948 (811) UBS Securities Australia Ltd Sell 6,615 (618) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 5,734 (581) UBS Securities Australia Ltd Sell 5,734 (581) UBS Securities Australia Ltd Sell 5,734 (72) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Austra	un-14	UBS Securities Australia Ltd	Sell	5,968	(609)	Ordinary
UBS Securities Australia Ltd Sell 10,270 (1,049) UBS Securities Australia Ltd Sell 1,566 (1,609) UBS Securities Australia Ltd Sell 578 (591) UBS Securities Australia Ltd Sell 578 (591) UBS Securities Australia Ltd Sell 11,1,44 (1,1,347) UBS Securities Australia Ltd Sell 293,282 (29,342) UBS Securities Australia Ltd Sell 7,948 (811) UBS Securities Australia Ltd Sell 5,912 (273) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 2,156 (575) UBS Securities Australia Ltd Sell 2,166 (127) UBS Securities Australia Ltd Sell 2,166 (129) UBS Securities A	un-14	UBS Securities Australia Ltd	Sell	10,251	(1,046)	Ordinary
UBS Securities Australia Ltd Sell 1,566 (1,049) UBS Securities Australia Ltd Sell 5,78 (59) UBS Securities Australia Ltd Sell 5,73 (59) UBS Securities Australia Ltd Sell 11,1,144 (11,377) UBS Securities Australia Ltd Sell 35,282 (59,42) UBS Securities Australia Ltd Sell 6,056 (618) UBS Securities Australia Ltd Sell 6,056 (618) UBS Securities Australia Ltd Sell 3,832 (618) UBS Securities Australia Ltd Sell 3,734 (81) UBS Securities Australia Ltd Sell 2,519 (757) UBS Securities Australia Ltd Sell 2,510 (152) UBS Securities Australia Ltd Sell 2,510 (152) UBS Securities Austral	un-14	UBS Securities Australia Ltd	Sell	3,045	(311)	Ordinary
UBS Securities Australia Ltd Sell 1,566 (160) UBS Securities Australia Ltd Sell 578 (595) UBS Securities Australia Ltd Sell 11,144 (11,347) UBS Securities Australia Ltd Sell 293,282 (394) UBS Securities Australia Ltd Sell 6,515 (618) UBS Securities Australia Ltd Sell 6,615 (618) UBS Securities Australia Ltd Sell 6,615 (618) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 2,519 (227) UBS Securities Australia Ltd Sell 2,519 (287) UBS Securities Australia Ltd Sell 2,519 (27) UBS Securities Australia Ltd Sell 2,519 (150) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (180) UBS Securities Australia Ltd Sell 2,340 (22) UBS Securities Australia L	un-14	UBS Securities Australia Ltd	Sell	10,270	(1,049)	Ordinary
UBS Securities Australia Ltd Sell 578 (59) UBS Securities Australia Ltd Sell 11,144 (1347) UBS Securities Australia Ltd Sell 293,282 (595) UBS Securities Australia Ltd Sell 293,282 (79,642) UBS Securities Australia Ltd Sell 7,948 (811) UBS Securities Australia Ltd Sell 6,615 (675) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 1,912 (195) UBS Securities Australia Ltd Sell 2,166 (577) UBS Securities Australia Ltd Sell 2,519 (277) UBS Securities Australia Ltd Sell 2,516 (400) UBS Securities Australia Ltd Sell 2,528 (595) UBS Securities Australia Ltd Sell 2,166 (120) UBS Securities Australia Ltd Sell 1,488 (122) UBS Securities Australia Ltd Sell 1,488 (123) UBS Securities Austral	un-14	UBS Securities Australia Ltd	Sell	1,566	(160)	Ordinary
UBS Securities Australia Ltd Sell 11,144 (1547) UBS Securities Australia Ltd Sell 11,144 (11,347) UBS Securities Australia Ltd Sell 35,382 (391) UBS Securities Australia Ltd Sell 6,055 (618) UBS Securities Australia Ltd Sell 6,055 (618) UBS Securities Australia Ltd Sell 1,301 (678) UBS Securities Australia Ltd Sell 1,301 (678) UBS Securities Australia Ltd Sell 2,519 (577) UBS Securities Australia Ltd Sell 3,746 (678) UBS Securities Australia Ltd Sell 3,746 (678) UBS Securities Australia Ltd Sell 3,746 (180) UBS Securities Australia Ltd Sell 3,746 (180) UBS Securities Australia Ltd Sell 1,488 (182) UBS Securities Australia Ltd Sell 1,488 (182) UBS Securities Australia Ltd Sell 2,340 (180) UBS Securities Austra	un-14	UBS Securities Australia Ltd	Seli	578	(65)	Ordinary
UBS Securities Australia Ltd Sell 111,144 (11,347) UBS Securities Australia Ltd Sell 3,538 (361) UBS Securities Australia Ltd Sell 7,948 (817) UBS Securities Australia Ltd Sell 6,615 (678) UBS Securities Australia Ltd Sell 6,615 (678) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 2,519 (277) UBS Securities Australia Ltd Sell 2,519 (277) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,468 (180) UBS Securities Australia Ltd Sell 1,468 (180) UBS Securities Australia Ltd Sell 1,468 (180) UBS Securities Australia Ltd Sell 1,748 (180) UBS Securities Australi	41-nu	UBS Securities Australia Ltd	Self	5,878	(595)	Ordinary
UBS Securities Australia Ltd Self 3,338 (157) UBS Securities Australia Ltd Self 29,428 (29,42) UBS Securities Australia Ltd Self 6,515 (618) UBS Securities Australia Ltd Self 6,515 (618) UBS Securities Australia Ltd Self 1,301 (133) UBS Securities Australia Ltd Self 2,166 (221) UBS Securities Australia Ltd Self 2,166 (221) UBS Securities Australia Ltd Self 2,166 (221) UBS Securities Australia Ltd Self 2,25 (58) UBS Securities Australia Ltd Self 2,166 (100) UBS Securities Australia Ltd Self 2,16 (100) UBS Securities Australia Ltd Self 2,16 (100) UBS Securities Australia Ltd Self 2,16 (100) UBS Securities Australia Ltd Self 2,340 (229) UBS Securities Australia Ltd Self 2,340 (239) UBS Securities Australia Ltd </td <td>un-14</td> <td>UBS Securities Australia Itd</td> <td>ق</td> <td>111 144</td> <td>(71 347)</td> <td>Ordinary</td>	un-14	UBS Securities Australia Itd	ق	111 144	(71 347)	Ordinary
UBS Securities Australia Ltd Sell 2.9.23 (2.9.942) UBS Securities Australia Ltd Sell 7.942 (811) UBS Securities Australia Ltd Sell 6.015 (6.18) UBS Securities Australia Ltd Sell 1,931 (813) UBS Securities Australia Ltd Sell 1,931 (813) UBS Securities Australia Ltd Sell 2,519 (221) UBS Securities Australia Ltd Sell 2,519 (221) UBS Securities Australia Ltd Sell 3,734 (881) UBS Securities Australia Ltd Sell 3,734 (881) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (182) UBS Securities Australia Ltd Sell 2,340 (229) UBS Securities Australia Ltd Sell 2,340 (180) UBS Securities Austral	17	I IBC Securities Australia 1-d	3	0000	(120)) d
UBS Securities Australia Ltd Sell 7,948 UBS Securities Australia Ltd Sell 6,615 (675) UBS Securities Australia Ltd Sell 6,615 (675) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 1,912 (195) UBS Securities Australia Ltd Sell 2,166 (221) UBS Securities Australia Ltd Sell 2,166 (227) UBS Securities Australia Ltd Sell 2,166 (288) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 1,468 (150) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 2,340 (229) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd Sell<		Upo Sacurities Abstralia Ltd	nac J	0,030	(195)	Ordinary
UBS Securities Australia Ltd Sell 6/15 UBS Securities Australia Ltd Sell 6/15 UBS Securities Australia Ltd Sell 8/25 UBS Securities Australia Ltd Sell 1,303 (133) UBS Securities Australia Ltd Sell 1,303 (133) UBS Securities Australia Ltd Sell 2,519 (257) UBS Securities Australia Ltd Sell 3,746 (381) UBS Securities Australia Ltd Sell 3,746 (381) UBS Securities Australia Ltd Sell 3,716 (382) UBS Securities Australia Ltd Sell 3,916 (400) UBS Securities Australia Ltd Sell 1,488 (150) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,689 (180) UBS Securities Australia Ltd Sell 1,780 (118) UBS Securities Australia Ltd Sell 2,340 (132) UBS Securities Australia Ltd Sell 2,740 (180) <td>41-14</td> <td>UBS Securities Australia Ltd</td> <td>Seil</td> <td>793,282</td> <td>(29,942)</td> <td>Ordinary</td>	41-14	UBS Securities Australia Ltd	Seil	793,282	(29,942)	Ordinary
UBS Securities Australia Ltd Sell 6615 (675) UBS Securities Australia Ltd Sell 6,056 (618) UBS Securities Australia Ltd Sell 1,911 (1951) UBS Securities Australia Ltd Sell 2,519 (237) UBS Securities Australia Ltd Sell 2,166 (227) UBS Securities Australia Ltd Sell 2,166 (227) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 5,829 (150) UBS Securities Australia Ltd Sell 1,468 (152) UBS Securities Australia Ltd Sell 1,468 (152) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,688 (182) UBS Securities Australia Ltd Sell 1,788 (18) UBS Securities Australia Ltd Sell 2,340 (229) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd<	un-14	UBS Securities Australia Ltd	Sell	7,948	(811)	Ordinary
UBS Securities Australia Ltd Sell 6,056 (618) UBS Securities Australia Ltd Sell 1,303 (331) UBS Securities Australia Ltd Sell 1,303 (133) UBS Securities Australia Ltd Sell 2,519 (227) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,689 (150) UBS Securities Australia Ltd Sell 1,789 (152) UBS Securities Australia Ltd Sell 2,340 (229) UBS Securities Australia Ltd Sell 2,340 (329) UBS Securities Australia Ltd	un-14	UBS Securities Australia Ltd	Seil	6,615	(675)	Ordinary
UBS Securities Australia Ltd Sell 3,832 (391) UBS Securities Australia Ltd Sell 1,911 (195) UBS Securities Australia Ltd Sell 2,519 (237) UBS Securities Australia Ltd Sell 2,166 (221) UBS Securities Australia Ltd Sell 3,734 (881) UBS Securities Australia Ltd Sell 6,252 (688) UBS Securities Australia Ltd Sell 3,916 (400) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,468 (122) UBS Securities Australia Ltd Sell 1,468 (123) UBS Securities Australia Ltd Sell 1,488 (132) UBS Securities Australia Ltd Sell 1,155 (118) UBS Securities Australia Ltd Sell 2,271 (239) UBS Securities Australia Ltd Sell 2,771 (239) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd	un-14	UBS Securities Australia Ltd	Sell	6.056	(618)	Ordinary
UBS Securities Australia Ltd Sell 1,371 (195) UBS Securities Australia Ltd Sell 1,303 (133) UBS Securities Australia Ltd Sell 2,516 (227) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 6,252 (689) UBS Securities Australia Ltd Sell 6,252 (689) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 215 (227) UBS Securities Australia Ltd Sell 215 (10) UBS Securities Australia Ltd Sell 2,340 (129) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd Sell 2,271 (395) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd	un-14	UBS Securities Australia Itd	los	3 832	(301)	racipio
UBS Securities Australia Ltd Sell 1,373 (133) UBS Securities Australia Ltd Sell 2,519 (257) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 5,828 (585) UBS Securities Australia Ltd Sell 3,916 (400) UBS Securities Australia Ltd Sell 1,488 (150) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,785 (118) UBS Securities Australia Ltd Sell 2,340 (132) UBS Securities Australia Ltd Sell 2,271 (222) UBS Securities Australia Ltd Sell 2,771 (239) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd	14	I IRS Sequities Australia 1td	روا	1011	(100)	Countrary
UBS Securities Australia Ltd Sell 1,502 (137) UBS Securities Australia Ltd Sell 2,166 (221) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 6,252 (589) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 1,468 (150) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,748 (152) UBS Securities Australia Ltd Sell 1,748 (182) UBS Securities Australia Ltd Sell 2,271 (229) UBS Securities Australia Ltd Sell 2,340 (18) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd<	1, 4	100 Committee Australia 14d	100	116,1	(06)	Ordinaly
UBS Securities Australia Ltd Sell 2,519 (257) UBS Securities Australia Ltd Sell 2,166 (231) UBS Securities Australia Ltd Sell 3,734 (381) UBS Securities Australia Ltd Sell 5,528 (589) UBS Securities Australia Ltd Sell 3,916 (400) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (152) UBS Securities Australia Ltd Sell 1,469 (152) UBS Securities Australia Ltd Sell 1,155 (172) UBS Securities Australia Ltd Sell 2,340 (229) UBS Securities Australia Ltd Sell 2,271 (222) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd		ODS Securities Australia Liu	1120	sus,1	(+33)	Ordinary
UBS Securities Australia Ltd Seli 3.746 (321) UBS Securities Australia Ltd Seli 3.734 (381) UBS Securities Australia Ltd Seli 5.828 (585) UBS Securities Australia Ltd Seli 1,469 (150) UBS Securities Australia Ltd Seli 1,469 (150) UBS Securities Australia Ltd Seli 1,489 (152) UBS Securities Australia Ltd Seli 1,688 (152) UBS Securities Australia Ltd Seli 2,340 (17) UBS Securities Australia Ltd Seli 2,340 (18) UBS Securities Australia Ltd Seli 2,71 (222) UBS Securities Australia Ltd Seli 2,71 (239) UBS Securities Australia Ltd Seli 5,828 (589) UBS Securities Australia Ltd Seli 5,828 (589) UBS Securities Australia Ltd Seli 4,151 (424)	- I-I-I	UBS Securities Australia Ltd	Jel.	2,519	(/57)	Ordinary
UBS Securities Australia Ltd Seli 3.734 (381) UBS Securities Australia Ltd Sell 6.252 (688) UBS Securities Australia Ltd Sell 5,628 (595) UBS Securities Australia Ltd Sell 1,660 (100) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd Sell 2,71 (329) UBS Securities Australia Ltd Sell 2,71 (329) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,629 UBS Securities Australia Ltd Sell 5,629 UBS Securities Australia Ltd Sell 4,151 <td>4</td> <td>UBS Securities Australia Ltd</td> <td>Self</td> <td>2,166</td> <td>(221)</td> <td>Ordinary</td>	4	UBS Securities Australia Ltd	Self	2,166	(221)	Ordinary
UBS Securities Australia Ltd Sell 6,252 (638) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,65 (152) UBS Securities Australia Ltd Sell 1,78 (172) UBS Securities Australia Ltd Sell 1,155 (178) UBS Securities Australia Ltd Sell 2,240 (229) UBS Securities Australia Ltd Sell 2,271 (222) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,752 (589) UBS Securities Australia Ltd Sell 5,752 (589) UBS Securities Australia Ltd Sell 4,151 (424)	un-14	UBS Securities Australia Ltd	Self	3,734	(381)	Ordinary
UBS Securities Australia Ltd Sell 58.28 (595) UBS Securities Australia Ltd Sell 3,916 (400) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,488 (122) UBS Securities Australia Ltd Sell 1,688 (132) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd Sell 2,71 (329) UBS Securities Australia Ltd Sell 2,71 (329) UBS Securities Australia Ltd Sell 5,828 (589) UBS Securities Australia Ltd Sell 5,620 (589) UBS Securities Australia Ltd Sell 5,620 (589) UBS Securities Australia Ltd Sell 5,620 (589) UBS Securities Australia Ltd Sell 4,151 (4,24)	un-14	UBS Securities Australia Ltd	Sell	6,252	(828)	Ordinary
UBS Securities Australia Ltd Sell 1,469 (400) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,469 (150) UBS Securities Australia Ltd Sell 1,488 (122) UBS Securities Australia Ltd Sell 1,06 (1) UBS Securities Australia Ltd Sell 2,271 (239) UBS Securities Australia Ltd Sell 2,271 (189) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,601 (694) UBS Securities Australia Ltd Sell 5,601 (694) UBS Securities Australia Ltd Sell 4,151 (4,14)	un-14	UBS Securities Australia Ltd	Sell	5,828	(262)	Ordinary
UBS. Securities Australia Ltd Sell 1,469 (1 50) UBS. Securities Australia Ltd Sell 215 (22) UBS. Securities Australia Ltd Sell 1,488 (152) UBS. Securities Australia Ltd Sell 2,340 (7) UBS. Securities Australia Ltd Sell 2,340 (739) UBS. Securities Australia Ltd Sell 2,271 (222) UBS. Securities Australia Ltd Sell 5,828 (595) UBS. Securities Australia Ltd Sell 5,621 (569) UBS. Securities Australia Ltd Sell 5,621 (580) UBS. Securities Australia Ltd Sell 4,151 (424)	un-14	UBS Securities Australia Ltd	Sell	3.916	(400)	Ordinary
UBS Securities Australia Ltd Sell 215 (22) UBS Securities Australia Ltd Sell 1,488 (152) UBS Securities Australia Ltd Sell 10 (11) UBS Securities Australia Ltd Sell 2,340 (239) UBS Securities Australia Ltd Sell 2,71 (239) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,762 (589) UBS Securities Australia Ltd Sell 4,151 (424)	14-14	LIBS Securities Australia Ltd	الع	1 460	(150)	Ordinan
UBS Securities Australia Ltd Sell 1,488 (12) UBS Securities Australia Ltd Sell 1,488 (13) UBS Securities Australia Ltd Sell 2,340 (29) UBS Securities Australia Ltd Sell 1,155 (18) UBS Securities Australia Ltd Sell 2,271 (232) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,680 (694) UBS Securities Australia Ltd Sell 5,680 (694) UBS Securities Australia Ltd Sell 4,151 (424)	10-14	LIBS Sequitities Australia Ltd	le J	216	(22)	Californ
UBS Securities Australia Ltd Sell 1,469 (132)	1,0	J. IBC Countition Australia Ltd	130	6077	(27)	Oldligiy Oldligiy
Ops Securities Australia Ltd Sell 10 (1) UBS Securities Australia Ltd Sell 2,340 (29) UBS Securities Australia Ltd Sell 1,155 (118) UBS Securities Australia Ltd Sell 2,271 (322) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 6,801 (691) UBS Securities Australia Ltd Sell 5,752 (588) UBS Securities Australia Ltd Sell 4,151 (424)		COUNTY AND COMMISSION	100	004,1	(75)	Ciginary
UBS Securities Australia Ltd Sell 2,340 (299) UBS Securities Australia Ltd Sell 1,155 (118) UBS Securities Australia Ltd Sell 2,271 (222) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,821 (595) UBS Securities Australia Ltd Sell 5,821 (595) UBS Securities Australia Ltd Sell 5,631 (594) UBS Securities Australia Ltd Sell 4,151 (424)	5	UBS Securities Australia Ltd	Sell	01	(i)	Ordinary
UBS Securities Australia Ltd Sell 1,155 (118) UBS Securities Australia Ltd Sell 2,271 (232) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 5,629 (594) UBS Securities Australia Ltd Sell 5,629 (594) UBS Securities Australia Ltd Sell 5,752 (598)	un-14	UBS Securities Australia Ltd	Sell	2,340	(239)	Ordinary
UBS Securities Australia Ltd Sell 2,271 (322) UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 6,801 (694) UBS Securities Australia Ltd Sell 5,762 (588) UBS Securities Australia Ltd Sell 4,151 (424)	un-14	UBS Securities Australia Ltd	Sell	1,155	(118)	Ordinary
UBS Securities Australia Ltd Sell 5,828 (595) UBS Securities Australia Ltd Sell 6,801 (694) UBS Securities Australia Ltd Sell 5,762 (588) UBS Securities Australia Ltd Sell 4,151 (4,24)	un-14	UBS Securities Australia Ltd	Sell	2,271	(232)	Ordinary
UBS Securities Australia Ltd Sell 6,801 (694) UBS Securities Australia Ltd Sell 5,762 (588) UBS Securities Australia Ltd Sell 4,151 (424)	un-14	UBS Securities Australia Ltd	Seil	5.828	(595)	Ordinary
UBS Securities Australia Ltd Sell 5,762 (588) UBS Securities Australia Ltd Sell 4,151 (424)	10-14	ILIBS Securities Australia I td	وا	6.801	(009)	Ordinan
UBS Securities Australia Ltd Sell 4,151 (424)	1,	I IDC Cocurition Australia 14d	les les	r3r 1	(1002)	Colonial
UBS Securities Australia Ltd (424)	:	COO Securities Australia Liu	120	20/10		
	V1-4.				(000)	, m

U5-JUN-14	UBS Securities Australia Ltd	- N	1,104	(115)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	9	1 027	(101)	Cac Cac
05-Jun-14	UBS Securities Australia Ltd	3	1 690	(176)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3.142	(327)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	4,128	(430)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	000'96	(10,000)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	2,880	(300)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,581	(373)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	3.	6,106	(989)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	2,016	(210)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	1,901	(198)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	1,440	(150)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	25,997	(2,708)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	4,128	(430)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Self	5,418	(292)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Seli	1,841	(192)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	2,897	(304)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	4,717	(495)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	88,762	(9,314)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	688'6	(086)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	3,297	(346)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	2,766	(605)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	4,670	(490)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	50,833	(5,334)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Seli	4,672	(490)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	16,200	(1,699)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	1,105	(116)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Jac Jac	14,007	(1,469)	Ordinary
05-Jun-14	LIDS Committee Australia Ltd	Sell Sell	797'1	(133)	Ordinary
05-lin-14	I IBS Securities Australia I th		2,754	(1,2,1)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	ي ا	44.143	(4 632)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	N See	877	(62)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	14,007	(1,469)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	ll-8c	6883	(1,037)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	4,117	(432)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	314	(33)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	143	(15)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Self	4,765	(200)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Jeg.	29,991	(3,147)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	2,106	(221)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	lles i	7,024	(737)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	a c	534	(95)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	la S	3,107	(326)	Ordinary
05-Jun-14	LIBS Societies Australia Ltd	lac o	2,012	(3/9)	Ordinary
05-lin-14	LIRC Securities Australia 14d	ķ 3	155 /	(700)	Ordinary
05-lin-14	LIBS Securities Australia Ltd	3	167,4	(248)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Se.	2,697	(283)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	lla _S	42,580	(4,468)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	5,842	(613)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	lleS	381	(40)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	16,200	(669'1)	Ordinary
	UBS Securities Australia Ltd	Sell	1,726	(181)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	61	(2)	Ordinary
l	UBS Securities Australia Ltd	Seli	190,600	(20,000)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	19,060	(2,000)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	58,352	(6,123)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	629	(99)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	Sell	19	(2)	Ordinary
05-Jun-14	UBS Securities Australia Ltd	lles.	14,571	(1,529)	Ordinary
06-Jun-14	UBS Securities Australia Ltd	Buy	21,485	2,226	Ordinary
06-JUN-14	UBS Securities Australia Ltd	Sell	36,262	(3,725)	Ordinary
06-Jun-14	UBS Securities Australia Ltd	Bliv	13.814	1,500	Ordinary
06-lun-14	UBS Securities Australia 1td	Bity	464 784	47.472	1 100
		2	100		7000

UBS Securities Australia Ltd Sell 1,693 UBS Securities Australia Ltd Sell 1,693 UBS Securities Australia Ltd Sell 3,733 UBS Securities Australia Ltd Sell 9,833 UBS Securities Australia Ltd Sell 9,833 UBS Securities Australia Ltd Sell 9,833 UBS Securities Australia Ltd Sell 3,736 UBS Securities Australia Ltd <	00-Jun-14					Oronioso
UBS Securities Australia Ltd Sell 1,80.3 UBS Securities Australia Ltd Sell 1,80.3 UBS Securities Australia Ltd Sell 1,30.3 UBS Securities Australia Ltd Sell 3,50.3 UBS Securities Australia Ltd Sell 3,50.3 UBS Securities Australia Ltd Sell 4,200,000 (5 UBS Securities Australia Ltd Sell 8,20.7 (4,58 UBS Securities Australia Ltd Sell 8,20.7 (4,58 UBS Securities Australia Ltd Sell 8,20.7 (4,58 UBS Securities Australia Ltd Sell 1,575.2 (4,58 UBS Securities Australia Ltd Sell 3,757.2 (4,58	06-Jun-14	UBS Securities Australia Ltd	. Zell	5,645	(576)	Ciditary
LOSS Securities Australia Lud Sell 4,214 LOSS Securities Australia Lud Sell 1,697 LOSS Securities Australia Lud Sell 5,733 LOSS Securities Australia Lud Sell 5,733 LOSS Securities Australia Lud Sell 2,739 LOSS Securities Australia Lud Sell 2,733 LOSS Securities Australia Lud Sell 2,733 LOSS Securities Australia Lud Sell 2,733 LOSS Securities Australia Lud Sell 3,758 LOSS Securities Australia	-Jun-14	UBS Securities Australia Ltd	Sell	1,803	(184)	Ordinary
UBS Securities Australia Ltd Sell 1,547 UBS Securities Australia Ltd Sell 3,939 UBS Securities Australia Ltd Sell 3,939 UBS Securities Australia Ltd Sell 3,939 UBS Securities Australia Ltd Sell 8,723 UBS Securities Australia Ltd Sell 1,058 UBS Securities Australia Ltd Sell 3,633 UBS Securities Australia Ltd Sell 3,734 UBS Securities Australia Ltd Sell 3,737 UBS Securities Australia Ltd <	-Jun-14	Obs Securities Australia Ltd	- T	7/7	(28)	Ordinary
UBS Securities Australia Ltd Sell 1,697 UBS Securities Australia Ltd Sell 5,733 UBS Securities Australia Ltd Sell 3,979 UBS Securities Australia Ltd Sell 3,733 UBS Securities Australia Ltd Sell 9,733 UBS Securities Australia Ltd Sell 8,733 UBS Securities Australia Ltd Sell 8,734 UBS Securities Australia Ltd Sell 1,056 UBS Securities Australia Ltd Sell 3,735 UBS Securities Australia Ltd Sell 3,735 UBS Securities Australia Ltd Sell 3,736 UBS Securities Australia Ltd Sell 3,737 UBS Securities Australia Ltd <	-Jun-14	UBS Securities Australia Ltd	Sell	4,214	(430)	Ordinary
UBS Securities Australia Ltd Sell 3/33 UBS Securities Australia Ltd Sell 3/33 UBS Securities Australia Ltd Sell 3/33 UBS Securities Australia Ltd Sell 4,500,000 55 UBS Securities Australia Ltd Sell 8/27 UBS Securities Australia Ltd Sell 8/27 UBS Securities Australia Ltd Sell 10,548 UBS Securities Australia Ltd Sell 1,058 UBS Securities Australia Ltd Sell 1,058 UBS Securities Australia Ltd Sell 3,873 UBS Securities Australia Ltd Sell 1,059 UBS Securities Australia Ltd Sell 3,785 UBS Securities Australia Ltd Sell 3,785 UBS Securities Australia Ltd Sell 3,775 UBS Securities Australi	-Jun-14	UBS Securities Australia Ltd	Sell	11,697	(1,193)	Ordinary
UBS Securities Autorials Ltd Sell 4,379 UBS Securities Autorials Ltd Sell 4,000,000 UBS Securities Autorials Ltd Sell 4,000,000 UBS Securities Autorials Ltd Sell 4,000,000 UBS Securities Autorials Ltd Sell 1,0752 UBS Securities Autorials Ltd Sell 3,1859 UBS Securities Autorials Ltd Sell 4,0727 UBS Securities Autorials Ltd Sell 1,075 UBS Securities Autorials Ltd Sell 1,050 UBS Securities Autorials Ltd Sell 1,050 UBS Securities Autorials Ltd Sell 1,050 UBS Securities Autorials Ltd Sell 3,174 UBS Securities Autorials Ltd Sell 3,175 UBS Securities Autorials Ltd Sell 3,176 UBS Securities Autorials Ltd Sell 3,176 UBS Securities Autorials	-Jul-14	Obs Securities Australia Ltd	18 J	5,733	(585)	Ordinary
UBS Securities Australia Ltd Sell 4,000,000 UBS Securities Australia Ltd Sell 9,273 UBS Securities Australia Ltd Sell 9,273 UBS Securities Australia Ltd Sell 9,273 UBS Securities Australia Ltd Sell 9,23 UBS Securities Australia Ltd Sell 10,648 UBS Securities Australia Ltd Sell 10,648 UBS Securities Australia Ltd Sell 37,837 UBS Securities Australia Ltd Sell 37,837 UBS Securities Australia Ltd Sell 37,839 UBS Securities Australia Ltd Sell 37,837 UBS Securities Australia Ltd Sell 37,837 UBS Securities Australia Ltd Sell 37,727 UBS Securities Australia Lt	1.00	on Securities Australia Ltd	liac li	5,5,5	(40p)	Ordinary
UBS Securities Australia Ltd Sell 47,000 CP UBS Securities Australia Ltd Sell 47,000 CP UBS Securities Australia Ltd Sell 67,273 CP UBS Securities Australia Ltd Sell 67,273 CP UBS Securities Australia Ltd Sell 67,273 CP UBS Securities Australia Ltd Sell 7,859 10,752 UBS Securities Australia Ltd Sell 37,859 10,752 UBS Securities Australia Ltd Sell 37,859 10,959 UBS Securities Australia Ltd Sell 37,750 10,959 UBS Securities Aust	11-14	Upo certifies Australia Ltd	nec.	5,503	(337)	Ordinary
UBS Securities Australia Ltd Sell 92.737 UBS Securities Australia Ltd Sell 9.273 UBS Securities Australia Ltd Sell 4.982 UBS Securities Australia Ltd Sell 4.982 UBS Securities Australia Ltd Sell 10,648 UBS Securities Australia Ltd Sell 47,873 UBS Securities Australia Ltd Sell 5,109 UBS Securities Australia Ltd Sell 5,109 UBS Securities Australia Ltd Sell 1,239 UBS Securities Australia Ltd Sell 3,756 UBS Securities Australia Ltd Sell 1,767 UBS Securities Australia Ltd Sell 3,706 UBS Securities Australia Ltd	1100	1 DC Counties Australia Ltd	Jen J	4,900,000	(500,000)	Ordinary
UBS Securities Australia Ltd Sell 5,731 UBS Securities Australia Ltd Sell 5,731 UBS Securities Australia Ltd Sell 10,752 UBS Securities Australia Ltd Sell 10,752 UBS Securities Australia Ltd Sell 10,752 UBS Securities Australia Ltd Sell 37,893 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 3,752 UBS Securities Australia Ltd Sell 3,753 UBS Securities Australia Ltd Sell 3,753 UBS Securities Australia Ltd Sell 3,753 UBS Securities Australia Ltd	1,01	I IBC Countries Australia ted	100	75,23	(70,0)	Ordinary
UBS Securities Australia Ltd Sell 4.78 UBS Securities Australia Ltd Sell 4.78 UBS Securities Australia Ltd Sell 10.752 UBS Securities Australia Ltd Sell 10.782 UBS Securities Australia Ltd Sell 37.893 UBS Securities Australia Ltd Sell 47.871 UBS Securities Australia Ltd Sell 47.871 UBS Securities Australia Ltd Sell 47.871 UBS Securities Australia Ltd Sell 5.109 UBS Securities Australia Ltd Sell 1.229 UBS Securities Australia Ltd Sell 1.239 UBS Securities Australia Ltd Sell 3.173 UBS Securities Australia Ltd Sell 3.173 UBS Securities Australia Ltd Sell 3.173 UBS Securities Australia Ltd Sell 3.153 UBS Securities Australia Ltd	17	I IBC Societies Australia Ltd	Jac J	7,12,6	(340)	Crainary
UBS Securities Australia Ltd Sell 4,09 UBS Securities Australia Ltd Sell 1,0548 UBS Securities Australia Ltd Sell 10,548 UBS Securities Australia Ltd Sell 38,534 UBS Securities Australia Ltd Sell 37,839 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 3,758 UBS Securities Australia Ltd	-lim-14	I IRS Securities Australia Itd	3	22,7,5	(00)	Ordinary
UBS Securities Australia Ltd Sell 10,628 UBS Securities Australia Ltd Sell 10,572 UBS Securities Australia Ltd Sell 37,829 UBS Securities Australia Ltd Sell 43,631 UBS Securities Australia Ltd Sell 1,239 UBS Securities Australia Ltd Sell 1,239 UBS Securities Australia Ltd Sell 1,239 UBS Securities Australia Ltd Sell 1,572 UBS Securities Australia Ltd Sell 1,757 UBS Securities Australia Ltd Sell 3,726 UBS Securities Australia Ltd Sell 3,736 UBS Securities Australia Ltd	-lin-14	LIBS Securities Australia 1td	3	7/08	(450)	Ordinary
UBS Securities Australia Ltd Seri 10,722 UBS Securities Australia Ltd Seri 10,723 UBS Securities Australia Ltd Seri 47,873 UBS Securities Australia Ltd Seri 47,873 UBS Securities Australia Ltd Seri 47,873 UBS Securities Australia Ltd Seri 40,727 UBS Securities Australia Ltd Seri 40,727 UBS Securities Australia Ltd Seri 1,030 UBS Securities Australia Ltd Seri 3,756 UBS Securities Australia Ltd Seri 2,508 UBS Securities Australia Ltd Seri 2,508 UBS Securities Australia Ltd	-lin-14	I IRS Securities Australia I td	B 3	10,540	(4001)	Ording
UBS Securities Australia Ltd Sell 38,534 UBS Securities Australia Ltd Sell 47,873 UBS Securities Australia Ltd Sell 37,869 UBS Securities Australia Ltd Sell 40,277 UBS Securities Australia Ltd Sell 40,277 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 1,030 UBS Securities Australia Ltd Sell 1,030 UBS Securities Australia Ltd Sell 1,465 UBS Securities Australia Ltd Sell 1,465 UBS Securities Australia Ltd Sell 33,734 UBS Securities Australia Ltd Sell 3,374 UBS Securities Australia Ltd Sell 3,176 UBS Securities Australia Ltd Sell 2,508 UBS Securities Australia Ltd	-fin-14	UBS Securities Australia 1td	3	10,040	(1,080)	Ordinary
UBS Securities Australia Ltd Sell 47.873 UBS Securities Australia Ltd Sell 47.873 UBS Securities Australia Ltd Sell 43.631 UBS Securities Australia Ltd Sell 43.631 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 1,290 UBS Securities Australia Ltd Sell 3,758 UBS Securities Australia Ltd Sell 3,778 UBS Securities Australia Ltd Sell 2,079 UBS Securities Australia Ltd Sell 2,070 UBS Securities Australia Ltd Sell 2,970 UBS Securities Australia Ltd	-kın-14	I IRS Securities Australia I to	, J	38 534	(060'1)	Ordinary
UBS Securities Australia Ltd Sell 37,859 UBS Securities Australia Ltd Sell 43,631 UBS Securities Australia Ltd Sell 5,109 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 3,375 UBS Securities Australia Ltd Sell 3,375 UBS Securities Australia Ltd Sell 3,375 UBS Securities Australia Ltd Sell 3,726 UBS Securities Australia Ltd Sell 2,908 UBS Securities Australia Ltd Sell 2,908 UBS Securities Australia Ltd	-kın-14	LIBS Securities Australia Ltd	, J	100,000	(A 905)	Ordinary
UBS Securities Australia Ltd Sell 43,631 UBS Securities Australia Ltd Sell 5,890 UBS Securities Australia Ltd Sell 1,990 UBS Securities Australia Ltd Sell 40,727 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 2,809 UBS Securities Australia Ltd Sell 33,725 UBS Securities Australia Ltd Sell 33,727 UBS Securities Australia Ltd Sell 33,727 UBS Securities Australia Ltd Sell 33,727 UBS Securities Australia Ltd Sell 33,709 UBS Securities Australia Ltd Sell 33,709 UBS Securities Australia Ltd Sell 33,709 UBS Securities Australia Ltd Sell 31,709 UBS Securities Australia Ltd Sell 31,709 UBS Securities Australia Ltd Sell 31,709 UBS Securities Australia Ltd Sell 3,709 UBS Securities Australia Ltd Sell 2,900,500 UBS Securities Australia Lt	- lin-14	I IRS Societities Australia 14d	ķ	37 950	(4,095)	Cidinary
UBS Securities Australia Ltd Self 5,1,990 UBS Securities Australia Ltd Sell 1,199 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 3,175 UBS Securities Australia Ltd Sell 3,174 UBS Securities Australia Ltd Sell 3,175 UBS Securities Australia Ltd Sell 2,509 UBS Securities Australia Ltd Sell 2,509 UBS Securities Australia Ltd Sell 2,509 UBS Securities Australia Ltd	-lun-14	UBS Securities Australia 1td	100	43 631	(2,0,5)	Ordinary
UBS Securities Australia Ltd Sell 17.29 UBS Securities Australia Ltd Sell 0.727 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 2,809 UBS Securities Australia Ltd Sell 3,725 UBS Securities Australia Ltd Sell 3,725 UBS Securities Australia Ltd Sell 3,725 UBS Securities Australia Ltd Sell 3,726 UBS Securities Australia Ltd Sell 3,736 UBS Securities Australia Ltd Sell 3,736 UBS Securities Australia Ltd Sell 2,908 UBS Securities Australia Ltd Sell 2,908 UBS Securities Australia Ltd Sell 2,908 UBS Securities Australia Ltd <	14 m-14	1 88 Securities Australia 14d	Coll	100,01	(000 3)	Ordinary
UBS Securities Australia Ltd Sell 40,717 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 2,109 UBS Securities Australia Ltd Sell 1,230 UBS Securities Australia Ltd Sell 1,233 UBS Securities Australia Ltd Sell 33,758 UBS Securities Australia Ltd Sell 33,758 UBS Securities Australia Ltd Sell 33,725 UBS Securities Australia Ltd Sell 3,726 UBS Securities Australia Ltd Sell 3,726 UBS Securities Australia Ltd Sell 3,727 UBS Securities Australia Ltd Sell 2,708,500 UBS Securities Australia Ltd Sell 2,708,500 UBS Securities Australia Ltd Sell 2,708,500 UBS Securities Australia Ltd Sell 2,710 UBS Securities Australia Lt	14	186 Securities Australia 14d	100	000	(000'0)	Ordinary
UBS Securities Australia Ltd Seli 2,109 UBS Securities Australia Ltd Sell 2,809 UBS Securities Australia Ltd Sell 1,030 UBS Securities Australia Ltd Sell 31758 UBS Securities Australia Ltd Sell 31758 UBS Securities Australia Ltd Sell 33,758 UBS Securities Australia Ltd Sell 31,74 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 31,750 UBS Securities Australia Ltd Sell 31,750 UBS Securities Australia Ltd Sell 31,750 UBS Securities Australia Ltd Sell 31,53 UBS Securities Australia Ltd Sell 3,500 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd	-lin-14	LIBS Securities Australia 14d	3	767 UV	(4)	Ordinary
UBS Securities Australia Ltd Sell 2,009 UBS Securities Australia Ltd Sell 1,030 UBS Securities Australia Ltd Sell 33,758 UBS Securities Australia Ltd Sell 33,758 UBS Securities Australia Ltd Sell 33,714 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 33,709 UBS Securities Australia Ltd Sell 3,709 UBS Securities Australia Ltd Sell 2,500 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Sell	2 109	(515)	Ordinary
UBS Securities Australia Ltd Seli 1,030 UBS Securities Australia Ltd Sell 1,233 UBS Securities Australia Ltd Sell 33,758 UBS Securities Australia Ltd Sell 33,758 UBS Securities Australia Ltd Sell 33,714 UBS Securities Australia Ltd Sell 33,779 UBS Securities Australia Ltd Sell 3,776 UBS Securities Australia Ltd Sell 3,770 UBS Securities Australia Ltd Sell 3,700 UBS Securities Australia Ltd Sell 2,700 UBS Securities Australia Ltd Sell 2,700 UBS Securities Australia Ltd Sell 3,500 UBS Securities Australia Ltd Sell 2,700 UBS Securities Australia Ltd Sell 2,710 UBS Securities Australia Ltd	5-Jun-14	UBS Securities Australia 11d	las	2 809	(080)	Ordinary
UBS Securities Australia Ltd Seil 1,293 UBS Securities Australia Ltd Seil 31,758 UBS Securities Australia Ltd Seil 31,758 UBS Securities Australia Ltd Seil 32,427 UBS Securities Australia Ltd Seil 33,174 UBS Securities Australia Ltd Seil 3,172 UBS Securities Australia Ltd Seil 3,176 UBS Securities Australia Ltd Seil 3,153 UBS Securities Australia Ltd Seil 3,153 UBS Securities Australia Ltd Seil 2,500 UBS Securities Australia Ltd Seil 2,1,107 UBS Securities Australia Ltd	5-Jun-14	UBS Securities Australia 11d	Sel	1.030	(106)	Ordinary
UBS Securities Australia Ltd Sell 33,758 UBS Securities Australia Ltd Sell 32,427 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 33,172 UBS Securities Australia Ltd Sell 33,173 UBS Securities Australia Ltd Sell 3,172 UBS Securities Australia Ltd Sell 3,173 UBS Securities Australia Ltd Sell 2,50 UBS Securities Australia Ltd Sell 2,90 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	J.	1 293	(133)	Ordinary
UBS Securities Australia Ltd Sell 515 UBS Securities Australia Ltd Sell 37,427 UBS Securities Australia Ltd Sell 39,174 UBS Securities Australia Ltd Sell 31,722 UBS Securities Australia Ltd Sell 31,722 UBS Securities Australia Ltd Sell 31,703 UBS Securities Australia Ltd Sell 31,709 UBS Securities Australia Ltd Sell 25,03 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 3,539 UBS Securities Australia Ltd Sell 2,308 UBS Securities Australia Ltd	5-Jun-14	UBS Securities Australia Ltd	Sell	33.758	(3 473)	Ordinary
UBS Securities Australia Ltd Sell 32,427 UBS Securities Australia Ltd Sell 1,465 UBS Securities Australia Ltd Sell 3,174 UBS Securities Australia Ltd Sell 3,176 UBS Securities Australia Ltd Sell 3,726 UBS Securities Australia Ltd Sell 3,956 UBS Securities Australia Ltd Sell 3,956 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,506,500 UBS Securities Australia Ltd Sell 2,71,107 UBS Securit	5-Jun-14	UBS Securities Australia Ltd	le.	515	(53)	Ordinary
UBS scorrites Australia Ltd Sell 1,465 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 31,702 UBS Securities Australia Ltd Sell 31,703 UBS Securities Australia Ltd Sell 31,703 UBS Securities Australia Ltd Sell 31,703 UBS Securities Australia Ltd Sell 31,503 UBS Securities Australia Ltd Sell 25,2 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd	5-Jun-14	UBS Securities Australia Ltd	Sell	32,427	(3,343)	Ordinary
UBS Securities Australia Ltd Sell 650 UBS Securities Australia Ltd Sell 650 UBS Securities Australia Ltd Sell 33,174 UBS Securities Australia Ltd Sell 3,726 UBS Securities Australia Ltd Sell 17,673 UBS Securities Australia Ltd Sell 3,996 UBS Securities Australia Ltd Sell 3,1709 UBS Securities Australia Ltd Sell 3,1709 UBS Securities Australia Ltd Sell 3,1709 UBS Securities Australia Ltd Sell 2,509 UBS Securities Australia Ltd Sell 2,509 UBS Securities Australia Ltd Sell 2,508 UBS Securities Australia Ltd	5-Jun-14	UBS Securities Australia Ltd	Sell	1,465	(151)	Ordinary
UBS securities Australia Ltd Sell 650 UBS Securities Australia Ltd Sell 3,174 UBS Securities Australia Ltd Sell 3,175 UBS Securities Australia Ltd Sell 1,1573 UBS Securities Australia Ltd Sell 3,1709 UBS Securities Australia Ltd Sell 3,1709 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,500 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 2,1,107 UBS Securities Australia Ltd Sell 2,23 UBS Securities Australia	5-Jun-14	UBS Securities Australia Ltd	Sell	39	(4)	Ordinary
UBS Securities Australia Ltd Sell 33.174 UBS Securities Australia Ltd Sell 2,726 UBS Securities Australia Ltd Sell 3,173 UBS Securities Australia Ltd Sell 3,296 UBS Securities Australia Ltd Sell 3,331 UBS Securities Australia Ltd Sell 3,133 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,52 UBS Securities Australia Ltd Sell 2,50 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,1107 UBS Securities Australia Ltd Sell 2,1107 UBS Securities Australia Ltd Sell 2,1107 UBS Securities Australia Ltd Sell 2,593 UBS Securities Australia Ltd Sell 2,593 UBS Securities Australia Ltd	5-Jun-14	UBS Securities Australia Ltd	Sell	029	(67)	Ordinary
UBS Securities Australia Ltd Sell 3,726 UBS Securities Australia Ltd Sell 17,673 UBS Securities Australia Ltd Sell 3,996 UBS Securities Australia Ltd Sell 3,709 UBS Securities Australia Ltd Sell 3,170 UBS Securities Australia Ltd Sell 3,170 UBS Securities Australia Ltd Sell 2,20 UBS Securities Australia Ltd Sell 2,20 UBS Securities Australia Ltd Sell 2,50 UBS Securities Australia Ltd Sell 2,70 UBS Securities Australia Ltd Sell <td>5-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>33,174</td> <td>(3,420)</td> <td>Ordinary</td>	5-Jun-14	UBS Securities Australia Ltd	Sell	33,174	(3,420)	Ordinary
UBS Securities Australia Ltd Sell 3,172 UBS Securities Australia Ltd Sell 3,296 UBS Securities Australia Ltd Sell 3,206 UBS Securities Australia Ltd Sell 3,709 UBS Securities Australia Ltd Sell 407 UBS Securities Australia Ltd Sell 2,52 UBS Securities Australia Ltd Sell 2,52 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,1,107 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd <td< td=""><td>5-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>2,726</td><td>(281)</td><td>Ordinary</td></td<>	5-Jun-14	UBS Securities Australia Ltd	Sell	2,726	(281)	Ordinary
UBS Securities Australia Ltd Sell 1,573 UBS Securities Australia Ltd Sell 3,996 UBS Securities Australia Ltd Sell 9,331 UBS Securities Australia Ltd Sell 9,331 UBS Securities Australia Ltd Sell 3,152 UBS Securities Australia Ltd Sell 252 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,1,107 UBS Securities Australia Ltd Sell 5,503 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 5,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd <	5-Jun-14	UBS Securities Australia Ltd	Sell	3,172	(327)	Ordinary
UBS Securities Australia Ltd Sell 3,996 UBS Securities Australia Ltd Sell 31,709 UBS Securities Australia Ltd Sell 10 UBS Securities Australia Ltd Sell 407 UBS Securities Australia Ltd Sell 25 UBS Securities Australia Ltd Sell 25 UBS Securities Australia Ltd Sell 25 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 2,91,107 UBS Securities Australia Ltd Sell 2,91 UBS Securities Australia Ltd Sell 2,79 UBS Securities Australia Ltd Sell 2,79 UBS Securities Australia Ltd Sell 2,79 UBS Securities Australia Ltd Sell 2,70 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Sell	17,673	(1,822)	Ordinary
UBS Securities Australia Ltd Sell 31,709 UBS Securities Australia Ltd Sell 9,331 UBS Securities Australia Ltd Sell 407 UBS Securities Australia Ltd Sell 252 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,603 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,710 UBS Securities Australia Ltd Sell 2,730 UBS Securities Australia Ltd	5-Jun-14	UBS Securities Australia Ltd	Sell	966'8	(412)	Ordinary
UBS Securities Australia Ltd Sell 9,331 UBS Securities Australia Ltd Sell 407 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,52 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 2,503 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,1107 UBS Securities Australia Ltd Sell 2,1107 UBS Securities Australia Ltd Sell 5,793 UBS Securities Australia Ltd Sell 2,273 UBS Securities Australia Ltd <t< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>31,709</td><td>(3,269)</td><td>Ordinary</td></t<>	-Jun-14	UBS Securities Australia Ltd	Sell	31,709	(3,269)	Ordinary
UBS Securities Australia Ltd Sell 10 UBS Securities Australia Ltd Sell 407 UBS Securities Australia Ltd Sell 252 UBS Securities Australia Ltd Sell 253 UBS Securities Australia Ltd Sell 6,063 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 2,110 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 5,78 UBS Securities Australia Ltd Sell 5,78 UBS Securities Australia Ltd Sell 5,78 UBS Securities Australia Ltd Sell 2,793 UBS Securities Australia Ltd Sell 2,793 UBS Securities Australia Ltd Sell 2,793 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Sell	155,9	(362)	Ordinary
UBS Securities Australia Ltd Sell 407 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 252 UBS Securities Australia Ltd Sell 2,006,500 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 3,579 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 21,107 UBS Securities Australia Ltd Sell 2,598 UBS Securities Australia Ltd Sell 2,598 UBS Securities Australia Ltd Sell 2,578 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Sell	10	(1)	Ordinary
UBS Securities Australia Ltd Sell 3.153 UBS Securities Australia Ltd Sell 252 UBS Securities Australia Ltd Sell C603 UBS Securities Australia Ltd Sell 7.080,500 UBS Securities Australia Ltd Sell 2.500,500 UBS Securities Australia Ltd Sell 2.500,500 UBS Securities Australia Ltd Sell 3.153 UBS Securities Australia Ltd Sell 3.153 UBS Securities Australia Ltd Sell 2.71,107 UBS Securities Australia Ltd Sell 2.1,107 UBS Securities Australia Ltd Sell 2.7,107 UBS Securities Australia Ltd Sell 2.7,107 UBS Securities Australia Ltd Sell 2.7,22 UBS Securities Australia Ltd Sell 3.153 UBS Securities Australia Ltd Sell 2.2,223 UBS Securities A	-Jun-14	UBS Securities Australia Ltd	Sell	407	(42)	Ordinary
UBS Securities Australia Ltd Sell 252 UBS Securities Australia Ltd Sell 23 UBS Securities Australia Ltd Sell 708 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 2,908,500 UBS Securities Australia Ltd Sell 3,579 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 21,107 UBS Securities Australia Ltd Sell 3,578 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 5,788 UBS Securities Australia Ltd Sell 5,788 UBS Securities Australia Ltd Sell 25,788 UBS Securities Australia Ltd Sell 25,100 UBS Securities Australia Ltd Sell 25,100 UBS Securities Australia Ltd Sell 22,210 UBS Securities Australia Ltd Sell 22,210 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Sell	3,153	(325)	Ordinary
UBS securities Australia Ltd Sell CDG3 UBS Securities Australia Ltd Sell C,063 UBS Securities Australia Ltd Sell C,308,500 UBS Securities Australia Ltd Sell C,303 UBS Securities Australia Ltd Sell C,510 UBS Securities Australia Ltd Sell C,511 UBS Securities Australia Ltd Sell C,110 UBS Securities Australia Ltd Sell C,107 UBS Securities Australia Ltd Sell C,528 UBS Securities Australia Ltd Sell C,522 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Sell	252	(56)	Ordinary
UBS Securities Australia Ltd Sell 6.063 UBS Securities Australia Ltd Sell 7.08 UBS Securities Australia Ltd Sell 2,5005,500 UBS Securities Australia Ltd Sell 1,484 UBS Securities Australia Ltd Sell 3,579 UBS Securities Australia Ltd Sell 2,91,107 UBS Securities Australia Ltd Sell 2,1,107 UBS Securities Australia Ltd Sell 2,1,107 UBS Securities Australia Ltd Sell 2,1,107 UBS Securities Australia Ltd Sell 5,78 UBS Securities Australia Ltd Sell 5,78 UBS Securities Australia Ltd Sell 5,78 UBS Securities Australia Ltd Sell 1,3,813 UBS Securities Australia Ltd Sell 22,100 UBS Securities Australia Ltd Sell 22,100 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 1,116 UBS Securities Australia Lt	-Jun-14	UBS Securities Australia Ltd	Sell	29	(3)	Ordinary
UBS Securities Australia Ltd Sell 708 UBS Securities Australia Ltd Sell 2,906,500 UBS Securities Australia Ltd Sell 2,508 UBS Securities Australia Ltd Sell 3,579 UBS Securities Australia Ltd Sell 2,91 UBS Securities Australia Ltd Sell 2,110 UBS Securities Australia Ltd Sell 2,113 UBS Securities Australia Ltd Sell 2,110 UBS Securities Australia Ltd Sell 2,578 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 2,578 UBS Securities Australia Ltd Sell 2,510 UBS Securities Australia Ltd Sell 2,210 UBS Securities Australia Ltd Sell 2,21 UBS Securities Australia Ltd Sell 2,21 UBS Securities Australia Ltd <t< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>690'9</td><td>(625)</td><td>Ordinary</td></t<>	-Jun-14	UBS Securities Australia Ltd	Sell	690'9	(625)	Ordinary
UBS Securities Australia Ltd Sell 2,908 500 (3) UBS Securities Australia Ltd Sell 3,573 1,844 UBS Securities Australia Ltd Sell 3,579 1,874 UBS Securities Australia Ltd Sell 3,579 1,910 UBS Securities Australia Ltd Sell 3,153 1,07 UBS Securities Australia Ltd Sell 7,933 1,107 UBS Securities Australia Ltd Sell 7,933 1,103 UBS Securities Australia Ltd Sell 5,578 1,03 UBS Securities Australia Ltd Sell 8,788 1,08 UBS Securities Australia Ltd Sell 8,788 1,00 UBS Securities Australia Ltd Sell 1,3,613 1,00 UBS Securities Australia Ltd Sell 29,100 1,00 UBS Securities Australia Ltd Sell 22,20 1,116 UBS Securities Australia Ltd Sell 22,223 1,116 UBS Securities Australia Ltd Sell 22,223 1,116 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	202	(73)	Ordinary
UBS Securities Australia Ltd Sell 2.503 UBS Securities Australia Ltd Sell 1,484 UBS Securities Australia Ltd Sell 3,579 UBS Securities Australia Ltd Sell 291 UBS Securities Australia Ltd Sell 21,107 UBS Securities Australia Ltd Sell 21,107 UBS Securities Australia Ltd Sell 7,993 UBS Securities Australia Ltd Sell 5,78 UBS Securities Australia Ltd Sell 8,788 UBS Securities Australia Ltd Sell 13,813 UBS Securities Australia Ltd Sell 13,813 UBS Securities Australia Ltd Sell 22,100 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	2,908,500	(300,000)	Ordinary
UBS Securities Australia Ltd Sell 1,484 UBS Securities Australia Ltd Sell 3,579 UBS Securities Australia Ltd Sell 291 UBS Securities Australia Ltd Sell 21,107 UBS Securities Australia Ltd Sell 27,107 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 2,578 UBS Securities Australia Ltd Sell 2,578 UBS Securities Australia Ltd Sell 2,310 UBS Securities Australia Ltd Sell 2,310 UBS Securities Australia Ltd Sell 2,310 UBS Securities Australia Ltd Sell 2,210 UBS Securities Australia Ltd Sell 2,210 UBS Securities Australia Ltd Sell 2,210 UBS Securities Australia Ltd Sell 2,21 UBS Securities Australia Ltd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	les	2,503	(258)	Ordinary
UBS Securities Australia Ltd Sell 3,579 UBS Securities Australia Ltd Sell 291 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 21,107 UBS Securities Australia Ltd Sell 7,93 UBS Securities Australia Ltd Sell 7,578 UBS Securities Australia Ltd Sell 8,788 UBS Securities Australia Ltd Sell 8,788 UBS Securities Australia Ltd Sell 13,613 UBS Securities Australia Ltd Sell 29,100 UBS Securities Australia Ltd Sell 22,200 UBS Securities Australia Ltd Sell 22,200 UBS Securities Australia Ltd Sell 22,200 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	1,484	(153)	Ordinary
UBS Securities Australia Ltd Sell 3.153 UBS Securities Australia Ltd Sell 3.153 UBS Securities Australia Ltd Sell 2.1,107 UBS Securities Australia Ltd Sell 7,993 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 8,788 UBS Securities Australia Ltd Sell 13,813 UBS Securities Australia Ltd Sell 29,100 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	3,579	(369)	Ordinary
UBS Securities Australia Ltd Sell 3.15.3 UBS Securities Australia Ltd Sell 23,100 UBS Securities Australia Ltd Sell 3,53 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 8,788 UBS Securities Australia Ltd Sell 13,813 UBS Securities Australia Ltd Sell 25,100 UBS Securities Australia Ltd Sell 22,100 UBS Securities Australia Ltd Sell 22,100 UBS Securities Australia Ltd Sell 22,100 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	291	(30)	Ordinary
UBS Securities Australia Ltd Sell 21,107 UBS Securities Australia Ltd Sell 3,153 UBS Securities Australia Ltd Sell 7,993 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 8,788 UBS Securities Australia Ltd Sell 8,788 UBS Securities Australia Ltd Sell 13,813 UBS Securities Australia Ltd Sell 29,100 UBS Securities Australia Ltd Sell 22,100 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 1,116	-Jun-14	'UBS Securities Australia Ltd	Sell	3,153	(325)	Ordinary
UBS Securities Australia Ltd Sell 7,993 UBS Securities Australia Ltd Sell 7,993 UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Self 8,788 UBS Securities Australia Ltd Self 13,813 UBS Securities Australia Ltd Self 29,100 UBS Securities Australia Ltd Self 22,100 UBS Securities Australia Ltd Self 22,223 UBS Securities Australia Ltd Self 22,223 UBS Securities Australia Ltd Self 7,156	-Jun-14	UBS Securities Australia Ltd	Sell	21,107	(2,176)	Ordinary
UBS Securities Australia Ltd Sell 5,798 UBS Securities Australia Ltd Self 5,578 UBS Securities Australia Ltd Self 8,788 UBS Securities Australia Ltd Self 13,813 UBS Securities Australia Ltd Self 29,100 UBS Securities Australia Ltd Self 29,100 UBS Securities Australia Ltd Self 22,123 UBS Securities Australia Ltd Self 22,223 UBS Securities Australia Ltd Self 22,223 UBS Securities Australia Ltd Self 1,116	Jun-14	UBS Securities Australia Ltd	Sell	3,153	(325)	Ordinary
UBS Securities Australia Ltd Sell 5,578 UBS Securities Australia Ltd Sell 252 UBS Securities Australia Ltd Sell 8,788 UBS Securities Australia Ltd Sell 13,813 UBS Securities Australia Ltd Sell 29,100 UBS Securities Australia Ltd Sell 22,203 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	7,993	(824)	Ordinary
UBS Securities Australia Ltd Seli 8.788 UBS Securities Australia Ltd Seli 8,788 UBS Securities Australia Ltd Seli 13,813 UBS Securities Australia Ltd Seli 29,100 UBS Securities Australia Ltd Seli 29,100 UBS Securities Australia Ltd Seli 22,223 UBS Securities Australia Ltd Seli 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	5,578	(575)	Ordinary
UBS Securities Australia Ltd Self 8,788 UBS Securities Australia Ltd Self 13,913 UBS Securities Australia Ltd Self 29,100 UBS Securities Australia Ltd Self 20,223 UBS Securities Australia Ltd Self 22,223 UBS Securities Australia Ltd Self 1,116	- Inn	UBS securities Australia Ltd	Sell	252	(26)	Ordinary
UBS Securities Australia Ltd Seli 13,813 UMS Securities Australia Ltd Sell 29,100 UBS Securities Australia Ltd Sell 8,022 UBS Securities Australia Ltd Sell 22,223 UBS Securities Australia Ltd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	8,788	(906)	Ordinary
UBS Securities Australia Ltd Sell 29,100 UBS Securities Australia Ltd Sell 8,022 UBS Securities Australia Ltd Sell 22,222 UBS Securities Australia Ltd Sell 1,116	Jun-14	UBS Securities Australia Ltd	Sell	13,813	(1,424)	Ordinary
UBS Securities Australia Itd Sell 8,022 UBS Securities Australia Itd Sell 22,223 UBS Securities Australia Itd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	Sell	29,100	(3,000)	Ordinary
UBS Securities Australia Ltd Sell 22,223	-Jun-14	UBS Securities Australia Ltd	lles.	8,022	(827)	Ordinary
UBS Securities Australia Ltd Sell 1,116	-Jun-14	UBS Securities Australia Ltd	Zell	22,223	(2,291)	Ordinary
	-Jun-14	UBS Securities Australia Ltd	Sell	1,116	(115)	Ordinary
UBS Securities Australia Ltd Sell 22,106	-Jun-14	ILIRS Securities Australia Ltd	= 0	20100		
1,610			100	77,106	(2,279)	Ordinary

J	Ţ	Ţ	Ţ	Ţ,	Ţ	Ţ	Ţ	Ţ]	Ţ	Ţ	Ţ	Ţ		1	Ţ.	Ţ			Ţ	T	Ţ			Ţ	Τ		Ţ	Ţ.	T	T		Ţ	Τ.	T.	Ī			Τ.	1.	1					Τ	T.	Τ			-
		1		Ordinary	Cidina	Ordinary		L	Ordinary	Ordinary	Ordinan	į	Ordinary		Ordinary			Н		Ordinary					Ordinary			Ordinary	1	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary			Ordinary	١		Ordinary															
(3,400)	(120)	(101)	(402)	(775 5)	(700/	(1098)	(798)	(513)	(397)	(405)	(061)	(1)	(429)	(162)	(2,584)	(430)	(2,617)	(150,000)	(595)	(400)	(2,108)	(296)	(1)	(174)	(4)(4)	(177)	(20)	(804)	(772)	(1,913)	(135)	(1,216)	(4,559)	(58)	(1,464)	(1,220)	(84)	(324)	(7)	ε	(772)	(18)	(716)	(523)	(381)	(1,869)	(426)	(381)	(41)	(420)	(325)	(462)	(800)	(5)	(1)
33,320	1 575	9/6/1	575 \$0	575,72	200,22	10 755	2,919	5,027	3,891	3,967	1,861	10	4,200	1,588	721.1	4.214	25,647	1,470,000	5,831	026,6	20,679	2,901	10	1,705	95,5	7,556	989	7,879	996'/	18,747	1,323	11,917	44,678	200,000	14,362	11,962	823	333	69	10	7,566	176	7,017	7,930	3,730	18,298	4,171	3,730	401	7,940 A 112	3.182	4,528	7,840	49	101
 Je Je	la Sell	lac	le le	la S	19	J	Sell	Set!	<u> </u>	38	Sell	Sell	Sell	Jac Jac	38	Sell	Sell	Sell	, J	Sell	Sell	Sell	Neil Neil	Sell	Sell	Sell	Sell	- J	Sell	Sell	Sell	Nell Nell	Se Se	Sell	Sell	Sell	Sell	Jac Jac	Sell	Sell	Sell	Sell	Sell	- B	Sell	les	Sell	Jes.	ğ						
 UBS Securities Australia Ltd	UBS Securities Australia Itd	1 BS Societies Australia 14d	PR Securities Australia Itd	UBS Securities Australia 1td	I IBS Securities Australia 1td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ind	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia (14	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	U8S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia ind
 06-Jun-14	06-lun-14	06-Jun-14	06-lin-14	06-lun-14	06-lim-14	06-Jun-14	06-Jun-14	ı	06-Jun-14		06-Jun-14	06-Jun-14	- 1	06-Jun-14	1	1	06-Jun-14	06-Jun-14	06-Jun-14	06-lin-14	06-Jun-14	06-Jun-14	06-Jun-14	06-Jun-14	06-Jun-14	06-Jun-14	1	- 1	06-Jun-14			06-Jun-14	1	1	T	06-Juli-14	T	Г	06-Jun-14		06-Jun-14	T	Ť	Ī	П	7	06-Jun-14								

10-Jun-14	מכן הביתוונים טחזוומום דום				
14-111-14-16-16-16-16-16-16-16-16-16-16-16-16-16-	1 105 Securities Australia Lid	la S	2,088	(065)	Ordinary
26-lin-14	LIRS Securities Australia Ind	Sell	960'C	(066)	Ordinary
1 n 1 1	LIBC Cocurities Australia Hd	100	47,730	(4,900)	Oldinary
1	OD Securities Australia Ltd	11-1	6	(7)	Collinary
06-101-14	I IBC Countries Australia 144	lio.	940/-	(1/1)	Ordinary
14-11-14	I IRC Counties Australia 144	100	7 145	(00//0)	O'dilliary
1 2 2 2	Con Control Action Co.	11-11	4,143	(430)	Ordinary
-101-14	Obs securities Australia Lid	Jac I	2 3	3	Ordinary
00-Jun-14	UBS Securities Australia Ltd	las	/9	3	Ordinary
-Jun-14	UBS Securities Australia Ltd	Self	25,209	(2,615)	Ordinary
06-Jun-14	UBS Securities Australia Ltd	Sell	29,884	(3,100)	Ordinary
06-Jun-14	UBS Securities Australia Ltd	Sell	569	(23)	Ordinary
06-Jun-14	UBS Securities Australia Ltd	Sell	1,350	(140)	Ordinary
06-Jun-14	UBS Securities Australia Ltd	Sell	9,486	(984)	Ordinary
D6-Jun-14	UBS Securities Australia Ltd		10	ε	Ordinary
5-Jun-14	UBS Securities Australia Ltd	Sell	10	(5)	Ordinary
-1:m-14	UBS Securities Australia I to	lla ₂	14 971	(1 553)	i i i
06-hm-14	I IRS Securities Australia I to	Sell	28.582	12 9651	Creding.
5-lin-14	1 IRS Sequities Australia 14d	100	3 173	(220)	Ordinary Ordinary
06-hin-14	Tibo Con witing Australia 144	100	366.68	(353)	Colonia
	CDD Securities Australia Ltd	III o	47,730	(4,900)	Orginary
00-JUIN-14	UBS Securities Australia Ltd	llex.	1/4/1	(4/7)	Ordinary
36-Jun-14	UBS Securities Australia Ltd	Sell	9,168	(951)	Ordinary
-Jun-14	UBS Securities Australia Ltd	Sell	47,236	(4,900)	Ordinary
36-Jun-14	UBS Securities Australia Ltd	Sell	47,236	(4,900)	Ordinary
06-Jun-14	UBS Securities Australia Ltd	Sell	10	(1)	Ordinary
06-Jun-14	UBS Securities Australia Ltd	Sell	29.547	(3.065)	Ordinary
36-Jun-14	UBS Securities Australia I td	100	1,648	(171)	Ordinany
06-1:0-14	I IBC Counties Australia 144	100	803.50	(000 0)	in in
1 1 1	The State August 200	li c	32,034	(000'C)	Oldinary
- Inc	ODS SECUTIVES AUSTRAINA LIG	Seil	900	(4)	Ordinary
10-Jun-14	UBS Securities Australia Ltd	Se	71,308	(7,284)	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	33,415	3,413	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Buy	8,226	842	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Buv	10.682	1001	Ordinary
0-lin-14	I IRS Securities Australia Ltd	Brity	5,610	723	Ordinan
0.17	LIBC Countition Americal and	600	11 753		Ordinary Ordinary
	De Consistent August Land	600	557,13	002,-	O Cilliary
0-2011-14	Obs securities Australia Ltd)ell	159,/3/	(10,303)	Ordinary
-Jun-14	UBS Securities Australia Ltd	Seil	960'09	(6,139)	Ordinary
0-Jun-14	UBS Global Asset Management	Buy	2,289	234	Ordinary
10-Jun-14	UBS Wealth Management Australia Ltd	Sell	19,042	(1.947)	Ordinary
0-lun-14	LiBS Wealth Management Australia Ltd	Transfer-In	VΛ	1 820	Ordinary
1 4 4	TIDE Commission Assessife to a	1101316111	C P	(20,1	Oldinary
-101-14	UBS Securities Austrana Ltd	Sell	67	(3)	Ordinary
-Jun-14	UBS Securities Australia Ltd	Stock borrowed	WA	20,000	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	24,831	2,559	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	127.471	13.120	Ordinary
1-hm-14	I IBS Securities Australia I to	Coll	72 613	(7 590)	Ordings
100	TIDE Commission Assessed and	100	235.0	(600,1)	Collinary
+1-nn1	UBS Securities Australia Ltd	Sell	5,755	(784)	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	6,684	689	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	485,335	20,000	Ordinary
-Jun-14	UBS Global Asset Management	Sell	405,096	(41,766)	Ordinary
1-Jun-14	UBS Wealth Management Australia Ltd	Sell	388	(40)	Ordinary
1-Jun-14	UBS Wealth Management Australia Ltd	Sell	42.762	(4.413)	Ordinan
1-km-14	URS Wealth Management Australia 11d	50	46 125	(6,874)	Ordinary
1 1 1 1	Late Management American Management of the Late of the		24,244	(120,0)	, o
1 .	ODS Weditt Management Australia Ltd)ell	447,42	(705,2)	Ordinary
1-Jun-14	UBS Wealth Management Australia Ltd	Buy	856'6	1,027	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	5,281,997	543,975	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	5,281,997	543,975	Ordinary
1-lun-14	UBS Securities Australia I to	Per v	6008 509	618 590	Ordina
1.0.1	LIDO Commission Augustical and	6	40.113	0.00	in in
1 1 1 1 1 4	Obsisted His Australia Liu	bay D. C.	49,113	0,030	Ciginary
Jun-14	UBS Securities Australia Ltd	Buy	69,465	7,154	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Sell	669	(72)	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,214	125	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2.606	577	Ordinary
1-lish-14	URS Securities Australia I to	Briv	11 035	1 136	Ordinary
1	Tipe Comision Australia and	600	000,1	2	A COUNTY
11-Jun-14	UBS Securities Australia Ltd	Buy	67/	7	Cadinary
-Jun-14	UBS Securities Australia Ltd	Buy	7,286	750	Ordinary

06-un-14 UBS Securities Australia Ltd 06-un-14 UBS Securities Australi	stralia Ltd stralia Ltd stralia Ltd stralia Ltd stralia Ltd stralia Ltd stralia Ltd		20,119 4,116 213,796	(2,087) (427) (22,178)	Ordinary Ordinary Ordinary Ordinary
	trala Ltd strala Ltd strala Ltd strala Ltd strala Ltd strala Ltd		213,796	(427)	Ordinary Ordinary Ordinary
	stalla Ltd stralia Ltd stralia Ltd stralia Ltd	* = ×	213,/96	(22,178)	Ordinary
	itralia Ltd tralia Ltd stralia Ltd			ú	Ordinary
	stralia Ltd stralia Ltd stralia I td		916	(CE)	,
	stralia Ltd	- X-	916	(36)	Ordinary
	stralia I td	Sell	1,957	(203)	Ordinary
		Sell	28	(9)	Ordinary
	stralia Ltd	Seff	3,692	(383)	Ordinary
	tralia Ltd	Sell	4,280	(444)	Ordinary
	stralia Ltd	Sell	23.502	(2.438)	Ordinary
	tralia I td	Call	900 90	(2007 c)	Ordinan
	train Ed	13	720,024	(2,7,00)	Oldinary
	יו פוום דום	lac J	100,024	(44,357)	Cidiriary
	itralia Ltd	Jell	26,028	(2,700)	Ordinary
	tralia Ltd	Sell	4,145	(430)	Ordinary
	tralia Ltd	Sell	9,042	(838)	Ordinary
1 1 1 1 1 1 1 1 1 1	tralia Ltd	Self	21.324	(2.212)	Ordinary
	tralia (1d	la y	364 402	(17,801)	Ordinan
	troils 14d	100	3007	(1001)	
	erolio Led	loc	2,000	(000)	Cidinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1	traila Ltd	361	2,230	(457)	Ordinary
	tralia Ltd	Sell	4,145	(430)	Ordinary
	tralia Ltd	Sell	20,340	(2,110)	Ordinary
	tralia Ltd	Sell	26,028	(2,700)	Ordinary
	tralia Ltd	Sell	26.028	(2.700)	Ordinary
	tralia Ltd	Sell	19	(2)	Ordinary
	tralia I td	وا	33 396	(TCA C)	Ordinan
	100	120	000,000	(+,7+,7)	y in any
	מפוס בוכ	100	100	(/ 1)	Clustrally
	tralia Ltd	Seli	4,145	(430)	Ordinary
	tralia Ltd	Sei	26,028	(2,700)	Ordinary
	tralia Ltd	Seli	26.028	(2.700)	Ordinary
Ш	tralia Ltd	Sell	4 145	(430)	Ordinary
П	trafia 1 td	Call	2 173	(000)	Ordinary
T	Day of the second		3/1/5	(676)	Clumary
	uana cro	Jell Sell	3,1,2	(329)	Ordinary
7	tralia Ltd	Sell	4,145	(430)	Ordinary
06-Jun-14 UBS Securities Aus	tralia Ltd	Sell	26,028	(2,700)	Ordinary
D6-Jun-14 UBS Securities Australia Ltd	tralia Ltd	Sell	26.028	(2.700)	Ordinary
Ī	tralia I td	9	3 173	(905)	Ordina
Ť	to diese	100	800 90	(000, 0)	, in the second
מסיים ביים ביים ביים ביים ביים ביים ביים	ון מוומ דונס	100	20,020	(7,700)	Ordinary
Ť	tralia Ltd	Seli	3,172	(329)	Ordinary
36-Jun-14 UBS Securities Australia Ltd	tralia Ltd	Sell	26,028	(2,700)	Ordinary
06-Jun-14 UBS Securities Australia Ltd	tralia Ltd	Sell	26.028	(0.700)	Ordinary
Ť	b+1 eiler+	Coll	900 90	(00/6)	Ordinary
Ť	Il alla Ltd	liac li	20,020	(2,700)	Ordinary
٦	tralia Ltd	Sell	2,/38	(584)	Ordinary
-Jun-14 UBS Securities Australia Ltd	tralia Ltd	- Sell	1,465	(152)	Ordinary
ř	tralia Ltd	Sell	1.947	(202)	Ordinary
t	il il il il il il il il il il il il il i	Coll	950.35	(202)	, and in a
Ť	tialia Ltd	Jell Jell	000,03	(2,701)	Cignary
	traila Ltd	Sell	12,908	(1,339)	Ordinary
06-Jun-14 UBS Securities Australia Ltd	tralia Ltd	Sell	5,871	(609)	Ordinary
ı	traffer 14d	Coll	200	(60)	Ordinan
1	tidalla LIU	251	/60	(66)	Cranary
06-Jun-14 UBS Securities Aus	tralia Ltd	Seli	1,003	(104)	Ordinary
	traija Ltd	Seti	1.003	(104)	Ordinary
Ī	tralia 11d	Sell	2 005	(auc/	Ordinary
District Continuos Coo	200		500/3	(203)	1
٦	tralia Ltd	Sell	15,424	(1,600)	Ordinary
06-Jun-14 UBS Securities Australia Ltd	tralia Ltd	Sell	20,311	(2,107)	Ordinary
Ī	train 14d	100	2 600	(000)) recipion
The state of the s	The second secon		6,002	(2007)	i i i
1	tralia Ltd	Zell Xell	. 48	(2)	Ordinary
06-Jun-14 UBS Securities Australia Ltd	tralia Ltd	Sell	2,805	(291)	Ordinary
OF-11 18 Securities Australia 11d	tralia I to	ll d'S	78 200	(5,000)	Ordinary
T	100	-	002,10	(000/0)	
JULE 14 IODS SECURITIES AUSTRALIA LIG	traile LIG	iàc	21,130	(5,2,2)	Ordinary
06-Jun-14 UBS Securities Aus	tralia Ltd	Sell	1,031	(107)	Ordinary
06-Jun-14 UBS Securities Australia Ltd	tralia Ltd	Sell	45.269	(4.696)	Ordinary
Ť	1		007 07	(4.000,000)	(in in in
7	ralia Ltd	Sell	40,488	(4,200)	Ordinary
	tralia Ltd	Sell	4,145	(430)	Ordinary
	rralia Ltd	Spill	40,488	(4,200)	Ordinary
T	P+I willed	وا	819 CV	(1CA A)	Ordinary,
T	traila Ltu	llas.	42,010	(4,4421)	Orginary
06-Jun-14 UBS Securities Australia Ltd	tralia Ltd	les.	24,113	(2,500)	Ordinary
Τ		= 5	700 5	, Jaron	

1-Jun-14 UBS Sec	UBS Securities Australia I td	Buy	12 591	1 208	Ordinary
Г	curities Australia Ltd	Buv	3,696	381	Ordinary
1-Jun-14 UBS Sec	UBS Securities Australia Ltd	Buy	39,246	4,046	Ordinary
П	JBS Securities Australia Ltd	Buy	11,388	1,174	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	20,012	2,061	Ordinary
	JBS Securities Australia Ltd	Buy	21,527	2,217	Ordinary
	UBS Securities Australia Ltd	Buy	17,760	1,829	Ordinary
11-Jun-14 UBS Sec	curities Australia Ltd	Buy	4,455	459	Ordinary
П	JBS Securities Australia Ltd	Buy	2,387	246	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	4,455	459	Ordinary
	UBS Securities Australia Ltd	Buy	9,705	1,000	Ordinary
	curities Australia Ltd	Buy	2,952	304	Ordinary
	curities Australia Ltd	Buy	3,146	324	Ordinary
	UBS Securities Australia Ltd	Buy	1,262	130	Ordinary
	UBS Securities Australia Ltd	Buy	2,796	288	Ordinary
UBS Sec	curities Australia Ltd	Buy	272	28	Ordinary
	curities Australia Ltd	Buy	2,224	529	Ordinary
	UBS Securities Australia Ltd	Buy	3,156	325	Ordinary
Ī	curities Australia Ltd	Buy	1,010	104	Ordinary
11-Jun-14 UBS Sec	UBS Securities Australia Ltd	Buy	1,583	163	Ordinary
	UBS Securities Australia Ltd	Buy	3,156	325	Ordinary
Ī	JBS Securities Australia Ltd	Buy	7,253	747	Ordinary
Ī	UBS Securities Australia Ltd	Buv	3.884	400	Ordinary
Ī	curities Australia Ltd	Buv	3.408	351	Ordinary
✝	UBS Securities Australia I td	Buy	3 700	381	Ordinary
11-Jun-14 UBS Sec	UBS Securities Australia Ltd	Buy	1.243	128	Ordinary
Γ	curities Australia Ltd	Buv	8.865	913	Ordinary
11-Jun-14 UBS Sec	UBS Securities Australia Ltd	Buv	15,662	1.613	Ordinary
ı	curities Australia Ltd	Buv	3.146	324	Ordinary
	UBS Securities Australia Ltd	Buy	4,535	467	Ordinary
ł	curities Australia Ltd	Buv	8.244	849	Ordinary
ı	JBS Securities Australia Ltd	Buv	2.136	220	Ordinary
UBS Sec	curities Australia Ltd	Buy	6.816	702	Ordinary
UBS Sec	UBS Securities Australia Ltd	Buv	3.146	324	Ordinary
UBS Sec	UBS Securities Australia Ltd	Buy	2,719	280	Ordinary
ı	curities Australia Ltd	Buy	2,622	270	Ordinary
1-Jun-14 UBS Sec	curities Australia Ltd	Buy	87,827	9,045	Ordinary
	JBS Securities Australia Ltd	Buy	952	86	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	1,515	156	Ordinary
_	JBS Securities Australia Ltd	Buy	2,330	240	Ordinary
1-Jun-14 UBS Sec	JBS Securities Australia Ltd	Buy	3,146	324	Ordinary
	curities Australia Ltd	Buy	1,573	162	Ordinary
-Jun-14 UBS Sec	JBS Securities Australia Ltd	Buy	5,923	610	Ordinary
Ē	JBS Securities Australia Ltd	Buy	10,681	1,100	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	19,401	1,998	Ordinary
	JBS Securities Australia Ltd	Buy	19,070	1,964	Ordinary
	UBS Securities Australia Ltd	Buy	1,72,71	234	Ordinary
1-Jun-14 UBS Sec	curities Australia Ltd	Buy	990'9	625	Ordinary
П	JBS Securities Australia Ltd	Buy	3,892	401	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	6,444	664	Ordinary
П	UBS Securities Australia Ltd	Buy	3,406	351	Ordinary
٦	JBS Securities Australia Ltd	Buy	1,892	195	Ordinary
	UBS Securities Australia Ltd	Buy	437	45	Ordinary
T	urities Australia Ltd	Buy	3,639	375	Ordinary
7	JBS Securities Australia Ltd	Buy	14,558	1,500	Ordinary
Ī	JBS Securities Australia Ltd	Buy	2,049	211	Ordinary
1-Jun-14 UBS Sec	Securities Australia Ltd	Buy	437	45.	Ordinary
11-Jun-14 UBS Sec	UBS Securities Australia Ltd	Buy	3,639	375	Ordinary
1	unties Australia Ltd	Buy	3,884	400	Ordinary
-Jun-14 UBS Sec	JBS Securities Australia Ltd	Buy	9,710	1,000	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	3,165	326	Ordinary
T	UBS Securities Australia Ltd	Buy	408	42	Ordinary
11-Jun-14 UBS Sec	UBS Securities Australia Ltd	Buy	427	44	Ordinary
	urities Australia Ltd	Buy	3,156	325	Ordinary
Г	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -				
	TILLING AUSTRALIA LTO	SUV	486	20	Ordinary

Г	П		П		7	_	7	Т	<u> </u>	Т	Т	Т	Т	Τ	П		Т	Т	Τ	Т	П	-	Т	Г	П		Т	Т	П	Т	1	Т	П	7	Т	Т	П	1	Т	Т	Т	П	1		Т	П	П	Т	Т	Т				Т	7
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Crdinary
494	289	138	176	400	1,500	2,000	702	1,634	068	369	9,00	375	824	324	45	25	330	324	184	409	438	73	35	325	325	324	324	324	585	325	325	350	2,000	423	196	324	325	369	324	000	136	3,400	400	324	869	3,367	23	136	26	1 8	325	1,178	778	896	707
4,802	2,809	1,341	1,710	3,888	14,580	19,420	6,816	15,866	8,642	3,583	952	3.156	8.001	3,146	437	202	3,204	3 146	1,787	3,971	4,253	709	340	3,156	3,156	3,146	3 156	3,146	2,680	3,156	3.156	3,399	19,420	4,107	3.146	3,146	3,156	3,583	3,146	4,633	1,321	33,014	3,884	3,145	6,778	32,694	223	1,903	0,000	913	3,156	11,427	7,547	3,590	
Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	gný.	Buy	and a	Buy	Bry	Buv	Buy	Buy	Buy	Buy	Buy.	Buy	Buy	Buy	Buy	Buv	Buy	Buy	Buy	Biry	Buy	Buy	Buy	Buv	Buy	Buy	Buy	Buv	Buy	Buy	Buy	Auy Pr.::	Buy	áng.														
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UDS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Securit	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UDS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	
11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-lun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14	11-Jun-14		- 1	1	11-Jun-14	11-Jun-14		1	1-Jun-14		П	11-Jun-14		1-3un-14													

-Jun-14	UBS Securities Australia Ltd	.≥.	3 802	202	Ordinan
1-Jun-14	UBS Securities Australia Ltd	Buy	3.356	346	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,959	202	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,232	127	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	6,480	899	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,125	116	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	446	46	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,045	417	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,067	110	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,317	342	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,317	445	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	8,546	881	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	5,364	553	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,677	276	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	11,640	1,200	Ordinary
un-14	UBS Securities Australia Ltd	Buy	5,927	611	Ordinary
un-14	UBS Securities Australia Ltd	Buy	4,093	422	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,387	143	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	962'9	089	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	728	75	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	35,085	3,617	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,948	407	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,386	246	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	42,932	4,426	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,162	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	3,162	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	89	7	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,075	317	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,162	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	28,411	2,929	Ordinary
1-JU-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
un-14	UBS Securities Australia Ltd	Buy	3,162	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	854	88	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
un-14	UBS Securities Australia Ltd	Buy	23,115	2,383	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,503	258	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3.143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Br	3.143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd) Ang	8 235	849	Ordinary
un-14	UBS Securities Australia Itd	Biv	3 143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	818	3 143	324	Ordinary
1-km-14	UBS Securities Australia 11d	818	3 162	328	Ordinan
1-hin-14	It IBS Securities Australia Ltd	Birx	4 365	AFO.	(in land
1-1 10-14	1 BC Counties Australia 14d	Buck	4,303	100	Ordinary
1-1 m-14	LIBC Counting Australia 14d	900	000 +	127	Colliary
	I IDC Committees Australia Ltd	eug.	006,1	134	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,143	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	340	35	Ordinary
	UBS Securities Australia Ltd	Buy	4,559	470	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	11,213	1,156	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	8,274	853	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	786	81	Ordinary
un-14	UBS Securities Australia Ltd	Buy	5,151	531	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	13,396	1,381	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,474	152	Ordinary
	UBS Securities Australia Ltd	8ny	2,697	278	Ordinary
	UBS Securities Australia Ltd	Buy	20.098	2 073	Ordinary
l		The same and the s		7/2/3	5
1 - Jun - 14	IUBS Securities Australia Ltd	Buv	27.5	2,072	Ordinary

1-km-14	LIBS Sportritips Australia Ltd	2	022	y	Ordinan
1-Jun-14	UBS Securities Australia I to	Bliv	308	17	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	6.535	673	Ordinary
1-lin-14	LIBS Securities Australia Ltd	Bin	3 106	324	Ording
1-lin-14	LIBS Securities Australia (1d	Bir	A 525	720	Ordinary
1-1	IBC Counties Australia 14d	100	64 053	2	Cicinary
11-4-014	I IBS Societies Australia I to	650	10035	1010	Ordinary
1-1.0-17	1 IBC Counties Australia (14	600	036	ניים	Oldinary
1 1 1	Lat. Grant A reference of the	60	505	000	Countries
100-	Obs Securities Australia Liu	end	4,502	443	Ordinary
4-14	UBS Securities Australia Ltd	Buy	320	33	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	418	43	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,146	324	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	7,671	790	Ordinary
Un-14	UBS Securities Australia Ltd	Buv	126	001	Ordinary
1-hm-14	LBS Securities Australia Ltd	Bin	124 754	12 848	Ordinan
1-hm-14	I IBS Societies Australia 144	, a	7669	2,27	Ordinary Ordinary
1.4.1	LIDE Commission Authority and	600	0,227	5 6	Oldinary
1 1 1	Upo Securities Adsirana Lid	Buy	500,0	200	Ordinary
41-10r-1	Securities Australia	gnà	3,140	324	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	13,594	1,400	Ordinary
un-14	UBS Securities Australia Ltd	Buy	612	63	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	28,703	2,956	Ordinary
un-14	UBS Securities Australia Ltd	Buy	3,146	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	36,898	3.800	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	3.301	340	Ordinary
1-lin-14	LIBS Securities Australia Ltd	, all	1 437	148	Ordenion
14	I IBS Securities Australia I to	6	0000	217	Ordinary
1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1	I IBS Cocuration Australia 144	600	0.00		Ordinary Ordinary
	Con percentage Australia and Property and Pr	(no	0	- [Countries
# 1	Ubb Securities Australia Ltd	éng	/74/	14/	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,641	169	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	320	33	Ordinary
lun-14	UBS Securities Australia Ltd	Buv	2.864	295	Ordinary
10-14	LIBS Securities Australia Ltd	Biv	8 244	840	Ordinan
1-1-114	I IBS Sacritias Australia I to	6	6 535	0.23	Ordina
	Language Australia Ltd	in a	626.00	2/0	Clumary
1-Jun-14	UBS Securities Australia Ltd	Buy	36,209	3,729	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	942	97	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	15,886	1,636	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	311	32	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3.146	324	Ordinary
un-14	LIBS Securities Australia 11d	Riv	3146	724	Ordinan
1-1-10-14	I IDC Contration Australia 14d	6.0	2,224	200	O'GILLIAN Y
1	COS Securities Australia city	âng.	7,224	677	Ordinary
41-14	UBS Securities Australia Ltd	Buy	3,146	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,330	240	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	1,359	140	Ordinary
un-14	UBS Securities Australia Ltd	Buv	505	25	Ordinary
11	I IDC Countition Ametralia (+4)	Bire	A 9 4E	100	Ordings.
:	ODS Securities Australia Liu	had	4,045	499	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	8,720	868	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,146	324	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Suv	3.165	326	Ordinary
11-14	I IRS Securities Australia I to	Brit	7 783	750	Ordinary
701	יייט פרייוויים איסיומוים בוס	ĥ	502.7	3 3	oralially Crallially
-In	UBS Securities Australia Lto	eny	767'9	548	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,156	325	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,156	325	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	1 998	206	Ordinary
1,0	I IDC Committee Australia 1 to	Din	000	36	, in the
	OD Consists Australia Ltd		045	3	Cidinaly
#1-Un	UBS Securities Australia Liu	Buy	17,460	1,800	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	291	30	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,221	525	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,259	336	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	3.405	351	Ordinary
1-jun-14	LIRS Securities Australia Itd	Bis	1135	117	Ordinary
:	UBS Securities Australia Lita	hna	001.0	/	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,580	592	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,423	456	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	7.721	962	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Biv	26.365	2718) To John
:	ממט שמשומים שמפון מוס	-			TOTO IN THE
	1 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			2,,10	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	1,474	152	Ordinary

1-Jun-14	UBS Securities Australia Ltd	Buy	17,024	1,755	Ordinary
11-lin-14	LIBS Securities Australia (14	Burk	127 31	1,143	Ordinary
-lin-14	I IRS Societies Australia Ltd	Bury	17470	050,	Ordinary
1.1.0.14	I IRC Sequirities Australia Hol	any Bulk	101,4	473	Ordinary
1-1:0-14	I IBC Countition Americal and	, and	764'6	300	Ordinary
1-110	LIBS Securities Australia 144	5.00	421,2	100.	Ordinary
11-110-14	I IRC Countries Australia 14d	Pung.	029'7	105,1	Ordinary
1-hm-14	LIRS Securities Australia 14d	B. W.	5//'1	501	Ordinary
1-hin-14	I IRS Securities Australia 14d	À C	3,002	310	Ordinary
1-jun-14	I IRS Securities Australia I to	B S	720		Ordinary
1-lin-14	USS Securities Australia Ltd	A A	207	13 0	Ordinary
1-Jun-14	UBS Securities Australia I td	Buy	2 852	700	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	36.55	767 E	Ordinary
1-lun-14	UBS Securities Australia 11d	B. c.	17 334	1 787	Ordinally
11-lun-14	ILIBS Securities Australia Itd	2 8	800.9	200	Ordinal y
11-Jun-14	UBS Securities Australia 11d	Sile of	6 567	729	Ordinary Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	2.086	215	ordina Ordina
11-Jun-14	UBS Securities Australia Ltd	Buv	20077	2 306	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4.947	510	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4.016	414	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	378	36	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,571	162	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	213	22	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,074	420	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,164	120	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	608'9	702	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	320	33	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	8,197	845	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	54,078	5,575	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	18,052	1,861	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Sell	11,733,786	(1,210,917)	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	9,351	362	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	31,454	3,236	Ordinary
1-Jun-14	UBS Securities Australia Ltd	8uy	2,916	300	Ordinary
+1-un	UBS Securities Australia Ltd	Buy	1,866	192	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	6,950	715	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,916	300	Ordinary
-Jun-14	UBS Securities Australia LTd	Buy	2,6/3	275	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,169	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	14,580	1,500	Ordinary
1-Jun-14	ubs Securities Australia Ltd	Buy	985,5	5/5	Ordinary
111111111111111111111111111111111111111	Class Securities Australia Lid	euy D.u.	11,464	081,1	Ordinary
1-Jun-14	Upo Securities Australia Ltd	BOJ	5,586	5/5	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	3,012	310	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	16	10	Ordinary
1-Jun-+4	UBS Securities Australia Ltd	acy	3,771	388	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,633	891	Ordinary
1.4 m. 14	LIDE Cocurition Australia Ltd	puy Dra:	14,580	000,1	Ordinary
1 1 1 1 1	I IBC Countries Australia Ltd	ouy D	16,271	4/0,1	Orginary
1-1-0-14	I IRC Countries Australia 14d	buy Buy	0,040	770	Ordinary
11-1 m-1/	IRC Sourities Australia Ltd	ony D	2,034	200	Ordinary
1 1	Libo Counties Australia Lib	buy buy	0000	770	Ordinary
1-Jun-14	Libs Securities Australia Ltd	lgn)	919,6	1,432	Ordinary
1-110-14	II IRS Societies Australia Ind	Puy Brix	5,103	370	Ordinary
1.0.14	II IRS Societies Australia Ind	pony	4,504	1000	Ordinary
1-hin-14	LIRS Securities Australia 11d	Buy	17/797	7,960	Ordinary
1-lun-14	LIBS Securities Australia Ltd	Briv	3 742	385	Ordinary
1-km-14	UBS Securities Australia Itd	Buy	54,75	573	Ordinary
1-hin-14	I IBS Securities Australia I td	B. D.	1 944	2000	Ordinary
1-lin-14	I IRS Securities Australia 14d	A Care	4 180	750	Ordinary
1-lin-14	I BS Securities Australia 14d	Si Si	1 750	100	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	757	45	Ordinary
		,			
4-01	185 Securities Australia 1th	,	32 557	2 2 40	Ordinary

UBS Securities Australia Ltd UBS Se
185 Securities Australia Ltd 186 Securities Australia Ltd 186 Securities A

11-Jun-14 11-Jun-14 11-Jun-14 11-Jun-14	UBS Securities Australia Ltd	and a	7,203	720	Conso
Jun-14 Jun-14 Jun-14		2	13,580	1.400	Ordinary
1-Jun-14 1-Jun-14	UBS Securities Australia Ltd	Buy	1,280	132	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	534	55	Ordinary
	UBS Securities Australia Ltd	Buv	28.945	2 984	Ordinary
Jun-14	UBS Securities Australia Ltd	Buv	8.720	899	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	446	46	Ordinary
Jun-14	UBS Securities Australia Ltd	Buv	3.162	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	2,803	289	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	58	9	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2.144	221	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3.647	376	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	786	81	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,162	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,996	412	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	21,864	2,254	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	223	23	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	398	41	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,862	295	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	3,162	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	136	14	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	5,151	531	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	8,643	891	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	252	56	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	834	98	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	4,935	209	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	111,395	11,484	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	7,581	782	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	436	45	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,636	375	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	14,543	1,500	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,348	139	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,871	296	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	166'6	1,030	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	25,724	2,652	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,300	134	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,162	326	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	12,203	1,258	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	6,518	672	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,520	466	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	4,142	427	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,483	256	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	060'8	834	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	378	39	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	20.894	2.154	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	650	29	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	6.412	661	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	4.831	498	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	8,245	850	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	5 892	809	Ordinan
1-Jun-14	UBS Securities Australia Ltd	Buv	4 409	455	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	5,165	533	Ordinary
Jun-14	UBS Securities Australia Itd	S. S.	213	22	Ording
1-Jun-14	UBS Securities Australia Ltd	Buy	4 651	480	Ordinary
11-Jun-14	IUBS Securities Australia Ltd	Suv	177	12	Ordinary
1-lun-14	I IRS Securities Australia 11d	Bits	6 512	27.2	Ordings
1-Jun-14	UBS Securities Australia Ltd	Riv	4 845	2002	Ordinary
l m-14	I IBS Securities Australia 14d	De la	6 703	200	Ordinary
1.lm.14	1 IRS Societies Australia 144	a a	201,03	90/	Ordinary
17	195 Constitut Australia Ltd	ond.	781	67	Ordinary
1 Jun 14	190 Contribut Australia Ltd	ána Arii	181,7	85	Ordinary
1	LIDE Contribes Australia Ltd	an)	4,496	404	Ordinary
-1	UBS Securities Australia Ltd	Buy	3,876	400	Ordinary
- 1	UBS Securities Australia Ltd	Buy	6,647	989	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	8 285	000	1

П	7	7	7		_	Γ	1	Т	T	_	_		Г	1	T	1	Т	7			Г	1			Г	Г	T	Т	T	1	1	_	_	_	_		, 	_	_			7	_		_	_	1							Γ-	_	_	_	_									1	т-
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Oldinary	Orginary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordioan	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary
385	2	295	374	1,679	127	5	61	5 6	0	9	80	99	115	3.300	306	270	370	840	3,578	282	111	107	(1,522,904)	(1,285,532)	791	704	110	4	69	S	141	365	726	125	175	233	124	373	906	5,302	5,794	200	74	326	326	981	898	297	412	75	326	326	240	14,212	83	77	326	400	8,726	2,631	244	2,641	2,000	326	149	14,588	383	326
3,742	61	2,867	3,635	16,320	1,234	49	593	202	/0/	28	778	642	1,118	32.076	3 160	601.0	9,169	8,165	34,778	2,741	1,079	1,040	14,756,940	12,456,805	7,673	6'859	1,067	39	699	485	1,368	3,541	7,042	1,213	1,698	2,260	1,203	3,618	7,818	51,429	26,202	4,848	718	3,162	3,162	9,516	8,420	2,881	3,996	728	3,162	3,162	2,328	137,856	802	747	3,162	3,880	84,642	25,521	2,367	25,618	006'29	3,162	1,445	141,504	3,715	3,162
Buy	Buy	Buy	Buy	Buy	Buy	BÇ	Buy		âng	Buy	Buy	Buy	Buy	Buy	Bir	ong.	park	Buy	Buy	Buy	Buy	Buy	Sell	Sell	Buy	Buy	Buy	Buy	Buy	Buy	Buv	Buy Buy	Buy	Buy	8uy	Buy	Buy	Buy	Buy	Buy	Buy	Виу	Buy	Buy	Buy	Buv	Buy																					
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IDC Consister Australia 14d	DO DECENTION AUSTRAIN LIG	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IRS Securities Australia 11d	One Securities Australia Liu	DOS SECULIOS AUSTRAIRA LIO	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	JBS Securities Australia Ltd	IBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd
П	T	Ť	1	11-Jun-14	11-Jun-14	11-Jun-14	Γ	ı	П	1 - Jun-14	- 1	11-Jun-14	11-Jun-14	11-Jun-14	T	T	T	1	٦				11-Jun-14		Ī	11-Jun-14	Г	Ī		Г	Γ	11-Jun-14		11-Jun-14	Ť	~	Ť	Ī	Ī	Ī	Ī	1	1	-1	11-hn-14		- 1		- 1	- 1	11-Jun-14	- 1	11-Jun-14 U	11-Jun-14 U		11-Jun-14 U				٦	11-Jun-14 U		11-Jun-14 ∪		ı	1 1	ı	11-Jun-14 U

	UBS Securities Australia Ltd	Buy	9,159	326	Ordinary
	Opp Securities Australia Ltd	-	~		
	UBS Securities Australia I td	Biv	7 075	775	Ordinary
	UBS Securities Australia Ltd	Buv	3 159	326	Ordinary
	IRS Securities Australia Ind	Bray	003	070	Ordinary Ordinary
	UBS Securities Australia Ltd	Buy	136	14	Ordinary
	UBS Securities Australia Ltd	Buy	921	95	Ordinary
Ħ	UBS Securities Australia Ltd	Buy	7,820	807	Ordinary
Ħ	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
İ	UBS Securities Australia Ltd	Buy	47,394	4,891	Ordinary
	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
П	UBS Securities Australia Ltd	Buy	5,727	591	Ordinary
	UBS Securities Australia Ltd	Buy	3,498	361	Ordinary
- 1	UBS Securities Australia Ltd	Buy	3,992	412	Ordinary
- 1	UBS Securities Australia Ltd	Buy	792	27	Ordinary
1-Jun-14	JBS Securities Australia Ltd	Buy	271	78	Ordinary
- 1	JBS Securities Australia Ltd	Buy	843	87	Ordinary
	JBS Securities Australia Ltd	Buy	28	9	Ordinary
	UBS Securities Australia Ltd	Buy	2,742	283	Ordinary
١	JBS Securities Australia Ltd	Buy	6,667	889	Ordinary
1	JBS Securities Australia Ltd	Buy	2,694	278	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	5,829	262	Ordinary
ı	JBS Securities Australia Ltd	Buy	2,907	300	Ordinary
1-Jun-14	JBS Securities Australía Ltd	Buy	10	1	Ordinary
ŀ	UBS Securities Australia Ltd	Buy	2,403	248	Ordinary
1-Jun-14 ∪	JBS Securities Australia Ltd	Buy	6,299	059	Ordinary
- 1	JBS Securities Australia Ltd	Buy	3,954	408	Ordinary
1-jun-14	UBS Securities Australia Ltd	Buy	242	25	Ordinary
- 1	JBS Securities Australia Ltd	Buy	5,853	604	Ordinary
- 1	JBS Securities Australia Ltd	Buy	30,456	3,143	Ordinary
٦	UBS Securities Australia Ltd	Buy	10	-1	Ordinary
┪	UBS Securities Australia Ltd	Buy	1,521	157	Ordinary
- 1	JBS Securities Australia Ltd	Buy	3,498	361	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	446	46	Ordinary
- 1	JBS Securities Australia Ltd	Buy	1,250	129	Ordinary
- 1	JBS Securities Australia Ltd	Buy	1,609	166	Ordinary
- 1	JBS Securities Australia Ltd	Buy	6,318	652	Ordinary
1	UBS Securities Australia Ltd	Buy	22,190	2,290	Ordinary
- 1	JBS Securities Australia Ltd	Buy	1,405	145	Ordinary
- 1	JBS Securities Australia Ltd	Buy	1,686	174	Ordinary
1-Jun-14 U	UBS Securities Australia Ltd	Buy	8,702	868	Ordinary
	JBS Securities Australia Ltd	Buy	2,578	366	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
į	JBS Securities Australia Ltd	Buy	3,275	338	Ordinary
1-Jun-14 U	UBS Securities Australia Ltd	Buy	1,328	137	Ordinary
1	UBS Securities Australia Ltd	Buy	22,539	2,326	Ordinary
l	JBS Securities Australia Ltd	Buv	12.016	1,240	Ordinary
ı	JBS Securities Australia Ltd	Buv	3.256	336	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buv	6.647	989	Ordinary
Ī	UBS Securities Australia Ltd	Buv	1,725	178	Ordinary
Γ	UBS Securities Australia Ltd	Buv	14,273	1.473	Ordinary
Ī	UBS Securities Australia Ltd	Buv	3.527	364	Ordinary
Γ	JBS Securities Australia Ltd	Buv	791	30	Ordinary
Ť	UBS Securities Australia ! td	Buy	1.550	160	Ordinary
T	IRS Securities Australia I to	Buy	3,605	377	ording)
Ť	IRS Securities Australia I to	Buy	357	3/2	Ordinary Ordinary
T	IBS Securities Australia 144	Biny	05/	2	Ciding,
ĺ	IRS Securities Australia I to	Riv	2 442	25.2	Ordinary Ordinary
1	LIBS Securities Australia Ltd	Bis	9699	109	Ordinary Special
1	IRS Securities Australia I th	B _I ×	1 269	151	Ordings
ı	IBS Societies Australia 14	Bury	602,1	77.6	Ordinary
T	IBS Securities Australia Itd	Bull	5,004	//7	Ordinary
T	IDS Committee Australia Ltd	Doug.	203 C	60	Ordinary
1-3411-14	Obsideration Australia Liu	Bay	2,507	607	Grainary
Т	DDS Secon lifes Australia Life	eny	2,300	957	Oralnary
11-Jun-14	UBS Securities Australia Ltd	Buy	6,560	677	Ordinary

(18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy 3,159 326 (18) Securities Australia Ltd Buy	URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby 3,150 URS Securities Australia Ltd Bby	1.4.0.14	I IDC Coo mitios Australia i tal	(Fig.	03	-	
UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159 326 UBS Securities Australia Lud Buy 3,159<	UBS Securities Australia Ltd BM 3.159 UBS Securities Australia Ltd BMy	n-14	UBS Securities Australia I td	Riv	3 140	328	Ordinary
May Securities Australia Ltd Buy 679 678 May Securities Australia Ltd Buy 73 759 726 May Securities Australia Ltd Buy 73 750 750 May Securities Australia Ltd Bu	URS Securities Australe Ltd Buy 519 URS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159 UNS Securities Australe Ltd Buy 3 159	Jn 14	UBS Securities Australia Ltd	Buy	3 159	326	Ordinary
LESS Securines Australia Ltd By Low 601 620	URS Securities Australia Ltd Buy 3 (195) URS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia	un-14	UBS Securities Australia I to	Riv	107	19	Ordinary
655 Sconfiles Autorales LM BM 3,159 3,65 655 Sconfiles Autorales LM BM 3,159 3,65 655 Sconfiles Autorales LM BM 3,159 3,56 655 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 650 Sconfiles Autorales LM BM 3,159 3,26 </td <td>UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy</td> <td>14</td> <td>I IRC Securities Australia I to</td> <td>Co a</td> <td>103</td> <td>5 6</td> <td>Ordinary Ordinary</td>	UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy	14	I IRC Securities Australia I to	Co a	103	5 6	Ordinary Ordinary
UBS Securities Australia Ltd BM 3,152 3,25 UBS Securities Australia Ltd BM 3,159 3,26 UBS Securities Australia Ltd BM 3,159 </td <td>UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia</td> <td>1</td> <td>I IDC Countries Australia 144</td> <td>, a</td> <td>031 6</td> <td>200</td> <td>Colliary</td>	UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia Ltd Buy 3 (195) UBS Securities Australia	1	I IDC Countries Australia 144	, a	031 6	200	Colliary
URS Securities Australia Ltd Buy 3,149 35 UBS Securities Australia Ltd Buy 3,159 35 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 <td>UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy</td> <td></td> <td>LIDE Counting Australia 14d</td> <td>(h)</td> <td>3,133</td> <td>250</td> <td>Collary</td>	UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 149 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy		LIDE Counting Australia 14d	(h)	3,133	250	Collary
UBS Securities Australia Ltd Buy 3,159 325 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 327 UBS Securities Australia Ltd Buy 3,159 327 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159<	UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	1	Upo Securities Australia Liu	, and	601'6	370	Ordinary
UBS Securities Australia Ltd Buy 4.15 4.25 4.25 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.30 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 4.20 1 UBS Securities Australia Ltd Buy 3,159 3.26 4.20 4.20 <td>UBS Securities Australia Ltd Buy 4,159 UBS Securities Australia Ltd Buy 4,159 UBS Securities Australia Ltd Buy 4,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy</td> <td>1 - 17</td> <td>Ubb Securities Australia Ltd</td> <td>Buy</td> <td>3,149</td> <td>325</td> <td>Ordinary</td>	UBS Securities Australia Ltd Buy 4,159 UBS Securities Australia Ltd Buy 4,159 UBS Securities Australia Ltd Buy 4,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	1 - 17	Ubb Securities Australia Ltd	Buy	3,149	325	Ordinary
UBS Securities Australia Ltd Buy 4.167 4.30 UBS Securities Australia Ltd Buy 3.199 3.26 UBS Securities Australia Ltd Buy 3.199 3.26 UBS Securities Australia Ltd Buy 3.199 3.25 UBS Securities Australia Ltd Buy 3.199 3.26 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 8,167 UBS Securities Australia Ltd Buy 8,148 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	- I - I 4	UBS Securities Australia Ltd	BUY	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199 326 UBS Securities Australia Ltd Buy 3,199<	UBS Securities Australia Ltd Buy 8,624 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	4,167	430	Ordinary
UBS Securities Australia Ltd Buy 3149 325 UBS Securities Australia Ltd Buy 3149 325 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3169 327 UBS Securities Australia Ltd Buy 3169 325 UBS Securities Australia Ltd Buy 3169 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 <th< td=""><td>UBS Securities Australia Ltd Buy 3149 UBS Securities Australia Ltd Buy 3159 UBS Securities Australia Ltd Buy 3159</td><td>41-14 14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>8,624</td><td>830</td><td>Ordinary</td></th<>	UBS Securities Australia Ltd Buy 3149 UBS Securities Australia Ltd Buy 3159	41-14 14	UBS Securities Australia Ltd	Buy	8,624	830	Ordinary
UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy 3,159 326 UBS Securities Australia Ltd Bluy	UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia Ltd Buy 3 (159) UBS Securities Australia	4-in	UBS Securities Australia Ltd	Buy	3,149	325	Ordinary
UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3169 327 UBS Securities Australia Ltd Buy 3169 328 UBS Securities Australia Ltd Buy 3169 326 UBS Securities Australia Ltd Buy 3169 <th< td=""><td>UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia</td><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,159</td><td>326</td><td>Ordinary</td></th<>	UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia	un-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3169 325 UBS Securities Australia Ltd Buy 3169 326 UBS Securities Australia Ltd Buy 3169 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 <th< td=""><td>UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy</td><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,159</td><td>326</td><td>Ordinary</td></th<>	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 3 (169 325 UBS Securities Australia Ltd Buy 3 (169 325 UBS Securities Australia Ltd Buy 3 (169 325 UBS Securities Australia Ltd Buy 3 (169 325 UBS Securities Australia Ltd Buy 3 (159 326 UBS Securities Australia Ltd Buy 3 (159 326 UBS Securities Australia Ltd Buy 3 (159 326 UBS Securities Australia Ltd Buy 3 (149 325 UBS Securities Australia Ltd Buy 3 (149 326 UBS Securities Australia Ltd Buy 3 (149 326 UBS Securities Australia Ltd Buy 3 (149 326 UBS Securities Australia Ltd Buy 3 (159 326 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 3 169 UBS Securities Australia Ltd Buy 3 169 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 3149 325 UBS Securities Australia Ltd Buy 3169 326 UBS Securities Australia Ltd Buy 3169 325 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3169 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 <th< td=""><td>UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy</td><td>1-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,169</td><td>327</td><td>Ordinary</td></th<>	UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	1-Jun-14	UBS Securities Australia Ltd	Buy	3,169	327	Ordinary
UBS Securities Australia Ltd Buy 3,159 325 UBS Securities Australia Ltd Buy 3,159 325 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159<	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3.149	325	Ordinary
UBS Securities Australia Ltd Bly 3,149 3.25 UBS Securities Australia Ltd Bly 3,149 3.25 UBS Securities Australia Ltd Bly 3,159 3.26 UBS Securities Australia Ltd Bly 3,159 3.26 UBS Securities Australia Ltd Bly 3,149 3.25 UBS Securities Australia Ltd Bly 3,159 3.26 UBS Securities Australia Ltd Bly	UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	41-00	UBS Securities Australia Ltd	Buy	3 159	326	Ordinary
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,149 325 UBS Securities Australia Ltd Buy 3,149 326 UBS Securities Australia Ltd Buy 3,149 326 UBS Securities Australia Ltd Buy 3,149 326 UBS Securities Australia Ltd Buy 3,159 3,26 UBS Securities Australia Ltd Buy 3,159	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	11-14	I IRS Securities Australia I td	Buy	3149	325	Ordinan
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,149 325 UBS Securities Australia Ltd Buy 3,149 326 UBS Securities Australia Ltd Buy 3,149 326 UBS Securities Australia Ltd Buy 3,149 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159<	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	10-14	I IBS Securities Australia I to	Biss	2 150	326	Ordinan
Construction of the control	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	11	P. IDC Constitute Australia 14d	(50	3 150	200	Ordinary.
UBS Securities Australia Ltd Buy 3,139 3.26 UBS Securities Australia Ltd Buy 3,149 3.26 UBS Securities Australia Ltd Buy 3,149 3.26 UBS Securities Australia Ltd Buy 3,149 3.25 UBS Securities Australia Ltd Buy 3,159 3.26 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy		Late of contract of contract of the contract o	900	60,10	320	Oldinal y
UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3149 325 UBS Securities Australia Ltd Buy 3149 325 UBS Securities Australia Ltd Buy 3149 326 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 <th< td=""><td>UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,169 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy</td><td></td><td>COO Securities Australia Liu</td><td>and .</td><td>901,0</td><td>270</td><td>Ordinary</td></th<>	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,169 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy		COO Securities Australia Liu	and .	901,0	270	Ordinary
UBS Securities Australia Ltd Bluy 3159 326 UBS Securities Australia Ltd Buy 3169 325 UBS Securities Australia Ltd Buy 3149 325 UBS Securities Australia Ltd Buy 3149 325 UBS Securities Australia Ltd Buy 3149 325 UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3159 <t< td=""><td>UBS Securities Australia Ltd Buy 3159 UBS Securities Australia Ltd Buy 3159 UBS Securities Australia Ltd Buy 3149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy <</td><td>4</td><td>UBS Securities Australia Ltd</td><td>Suy</td><td>3,159</td><td>326</td><td>Ordinary</td></t<>	UBS Securities Australia Ltd Buy 3159 UBS Securities Australia Ltd Buy 3159 UBS Securities Australia Ltd Buy 3149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy <	4	UBS Securities Australia Ltd	Suy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,149 325 UBS Securities Australia Ltd Buy 3,149 325 UBS Securities Australia Ltd Buy 3,159 325 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159<	UBS Securities Australia Ltd Buy 3.159 UBS Securities Australia Ltd Buy 3.149 UBS Securities Australia Ltd Buy 3.149 UBS Securities Australia Ltd Buy 3.149 UBS Securities Australia Ltd Buy 3.159 UBS Securities Australia Ltd Buy	4 - 14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Bby 3,207 331 UBS Securities Australia Ltd Bby 3,149 325 UBS Securities Australia Ltd Bby 3,149 325 UBS Securities Australia Ltd Bby 3,159 326 UBS Securities Australia Ltd Bby 3,159 326 UBS Securities Australia Ltd Buy 5,756 594 52 UBS Securities Australia Ltd Buy 3,159 326 326	UBS Securities Australia Ltd Buy 3,207 UBS Securities Australia Ltd Buy 3,207 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 3,149 325 UBS Securities Australia Ltd Buy 3,149 325 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 5,04 3,26 UBS Securities Australia Ltd Buy 3,159 3,26 UBS Securities Australia Ltd Buy 3,081 3,081 3,081 UBS Securities Australia Ltd Buy 5,040 2,02 2,17 UBS Securities Australia Ltd Buy 3,081 3,081 3,081 3,081 <td>UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,222 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,222 UBS Securities Australia Ltd Buy</td> <td>un-14</td> <td>UBS Securities Australia Ltd</td> <td>Влу</td> <td>3,207</td> <td>331</td> <td>Ordinary</td>	UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,222 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,222 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Влу	3,207	331	Ordinary
UBS Securities Australia Ltd Buy 31.49 325 UBS Securities Australia Ltd Buy 3,149 325 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 5,756 5,24 UBS Securities Australia Ltd Buy 3,159 3,26 UBS Securities Australia Ltd Buy <th< td=""><td>UBS Securities Australia Ltd Buy 3.149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy</td><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,149</td><td>325</td><td>Ordinary</td></th<>	UBS Securities Australia Ltd Buy 3.149 UBS Securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3,149	325	Ordinary
UBS Securities Australia Ltd Bby 3,149 325 UBS Securities Australia Ltd Bby 3,159 326 UBS Securities Australia Ltd Bby 5,756 594 UBS Securities Australia Ltd Bby 5,756 524 UBS Securities Australia Ltd Bby 5,756 3,69 UBS Securities Australia Ltd Bby 3,159 3,75 UBS Securities Australia Ltd Bby 2,432 2,62 UBS Securities Australia Ltd Bby <th< td=""><td>UBS securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,064 UBS Securities Australia Ltd Buy</td><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,149</td><td>325</td><td>Ordinary</td></th<>	UBS securities Australia Ltd Buy 3,149 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,064 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3,149	325	Ordinary
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 5,756 594 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 6,262 271 UBS Securities Australia Ltd Buy 6,122<	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 5,756 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,150 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 3,150 UBS Securities Australia Ltd Buy 3,150 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3,149	325	Ordinary
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 8,734 8,600 UBS Securities Australia Ltd Buy 8,735 5,26 UBS Securities Australia Ltd Buy 8,736 5,24 UBS Securities Australia Ltd Buy 1,5126 1,50 UBS Securities Australia Ltd Buy 1,5126 1,50 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 2,229 2,626 UBS Securities Australia Ltd Buy 2,122 2,024 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 5,756 UBS Securities Australia Ltd Buy 5,756 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,624 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buv	3.159	326	Ordinary
USS Securities Australia Ltd Buy 3,159 376 UBS Securities Australia Ltd Buy 3,159 376 UBS Securities Australia Ltd Buy 3,159 376 UBS Securities Australia Ltd Buy 3,159 376 UBS Securities Australia Ltd Buy 3,159 376 UBS Securities Australia Ltd Buy 3,159 376 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 1,996 206 UBS Securities Australia Ltd Buy 3,159 3,78 UBS Securities Australia Ltd Buy 3,081 3,78 UBS Securities Australia Ltd Buy 2,025 2,71 UBS Securities Australia Ltd Buy 2,626 2,71 UBS Securities Australia Ltd Buy 2,723 2,626 2,72 UBS Securities Australia Ltd Buy 2,723 2,93 2,93 UBS Securities Austral	UBS Securities Australia Ltd Buy 3,155 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,756 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 9,981 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 2,632 UBS Securities Australia Ltd Buy 2,732 UBS Securities Australia Ltd Buy	14	I IRS Securities Australia Ltd	Broy	3 150	375	Ordinan
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 5,756 594 UBS Securities Australia Ltd Buy 5,756 594 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 2,229 2,229 UBS Securities Australia Ltd Buy 3,159 3,66 UBS Securities Australia Ltd Buy 2,1	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 8,756 UBS Securities Australia Ltd Buy 5,756 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,156 UBS Securities Australia Ltd Buy 1,966 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,061 UBS Securities Australia Ltd Buy 3,061 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,239 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy	1,1	d IRC Committee Australia 14d	600	0.150	326	Coding
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 5,756 594 UBS Securities Australia Ltd Buy 5,756 594 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 2,420 66 UBS Securities Australia Ltd Buy 2,229 251 UBS Securities Australia Ltd Buy 6,173 668 UBS Securities Australia Ltd Buy 2,729 2,329 UBS Securities Australia Ltd Buy 2,732	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 5,756 UBS Securities Australia Ltd Buy 83,334 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,759 UBS Securities Australia Ltd Buy		1 DC Counties Australia Ltd	ord,	661,6	025	Ordinary
UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 8,756 594 502 UBS Securities Australia Ltd Buy 8,1334 8,600 UBS Securities Australia Ltd Buy 1,356 326 UBS Securities Australia Ltd Buy 1,356 1,561 UBS Securities Australia Ltd Buy 1,956 1,050 UBS Securities Australia Ltd Buy 1,956 206 UBS Securities Australia Ltd Buy 3,081 3,081 3,081 UBS Securities Australia Ltd Buy 2,626 27,1 UBS Securities Australia Ltd Buy 2,626 27,2 UBS Securities Australia Ltd Buy 2,626 236 UBS Securities A	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,156 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 5,756 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,620 UBS Securities Australia Ltd Buy 2,620 UBS Securities Australia Ltd Buy 2,632 UBS Securities Australia Ltd Buy 2,732 UBS Securities Australia Ltd Buy	1 :	ODS Seculities Australia LIC	Buy	8C1.C	070	Ordinary
UBS Securities Australia Ltd Buy 5756 534 UBS Securities Australia Ltd Buy 5.756 524 UBS Securities Australia Ltd Buy 5.756 524 UBS Securities Australia Ltd Buy 3,159 3.756 UBS Securities Australia Ltd Buy 15,126 1,561 UBS Securities Australia Ltd Buy 1,5126 1,500 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 2,1229 5,938 UBS Securities Australia Ltd Buy 2,1229 5,938 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 5159 UBS Securities Australia Ltd Buy 5.756 UBS Securities Australia Ltd Buy 5.756 UBS Securities Australia Ltd Buy 3.159 UBS Securities Australia Ltd Buy 1.296 UBS Securities Australia Ltd Buy 1.296 UBS Securities Australia Ltd Buy 3.159 UBS Securities Australia Ltd Buy 3.159 UBS Securities Australia Ltd Buy 3.159 UBS Securities Australia Ltd Buy 2.6.6 UBS Securities Australia Ltd Buy 3.159 UBS Securities Australia Ltd Buy 2.0.6 UBS Securities Australia Ltd Buy	41-I	UBS Securities Australia Ltd	eny	3,159	975	Ordinary
UBS Securities Australia Ltd Buy S5756 S94 UBS Securities Australia Ltd Buy 83,334 8,600 UBS Securities Australia Ltd Buy 5,756 326 UBS Securities Australia Ltd Buy 5,126 1,561 UBS Securities Australia Ltd Buy 1,5126 1,561 UBS Securities Australia Ltd Buy 9,981 1,030 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,081 318 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,1628 271 UBS Securities Australia Ltd Buy 2,732 2,30 UBS Securities Australia Ltd Buy 2,732 2,53 UBS Securities Australia Ltd Buy 2,732 2,53 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 8,756 UBS Securities Australia Ltd Buy 8,734 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 1,916 UBS Securities Australia Ltd Buy 1,956 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,081 UBS Securities Australia Ltd Buy 3,081 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 2,627 UBS Securities Australia Ltd Buy 2,628 UBS Securities Australia Ltd Buy 2,739 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,529 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 83.34 8.60 UBS Securities Australia Ltd Buy 504 8.52 UBS Securities Australia Ltd Buy 3,159 3.75 UBS Securities Australia Ltd Buy 1,396 1,561 UBS Securities Australia Ltd Buy 1,586 206 UBS Securities Australia Ltd Buy 3,159 3.76 UBS Securities Australia Ltd Buy 3,189 3.66 UBS Securities Australia Ltd Buy 3,189 3.76 UBS Securities Australia Ltd Buy 2,626 2.71 UBS Securities Australia Ltd Buy 2,626 2.73 UBS Securities Australia Ltd Buy 2,626 2.73 UBS Securities Australia Ltd Buy 2,632 5,938 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 83.34 UBS Securities Australia Ltd Buy 3.136 UBS Securities Australia Ltd Buy 3.156 UBS Securities Australia Ltd Buy 1,986 UBS Securities Australia Ltd Buy 1,986 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 8,00 UBS Securities Australia Ltd Buy 8,72 UBS Securities Australia Ltd Buy 8,73 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 5,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy <	UD-14	UBS Securities Australia Ltd	Buy	5,756	594	Ordinary
UBS Securities Australia Ltd Buy 504 52 UBS Securities Australia Ltd Buy 3,159 3,56 UBS Securities Australia Ltd Buy 1,5126 1,561 UBS Securities Australia Ltd Buy 9,981 1,030 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 2,42 66 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,473 680 UBS Securities Australia Ltd Buy 2,526 5,38 UBS Securities Australia Ltd Buy 2,122 2,04 66 UBS Securities Australia Ltd Buy 2,132 2,63 69 UBS Securities Australia Ltd<	UBS Securities Australia Ltd Buy 504 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 15,136 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 2,624 UBS Securities Australia Ltd Buy 2,624 UBS Securities Australia Ltd Buy 2,624 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	83,334	8,600	Ordinary
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 1,5126 1,561 UBS Securities Australia Ltd Buy 3,159 3,00 UBS Securities Australia Ltd Buy 3,159 3,66 UBS Securities Australia Ltd Buy 3,081 3,081 UBS Securities Australia Ltd Buy 3,081 3,081 UBS Securities Australia Ltd Buy 2,626 2,71 UBS Securities Australia Ltd Buy 6,628 6,380 UBS Securities Australia Ltd Buy 2,054 2,12 UBS Securities Australia Ltd Buy 2,632 2,51 UBS Securities Australia Ltd Buy 5,539 5,938 UBS Securities Australia Ltd Buy 5,739 5,938 UBS Securities Australia Ltd Buy <td>UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 1,126 UBS Securities Australia Ltd Buy 1,296 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 2,629 UBS Securities Australia Ltd Buy 2,632 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy</td> <td>un-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>204</td> <td>25</td> <td>Ordinary</td>	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 1,126 UBS Securities Australia Ltd Buy 1,296 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 2,629 UBS Securities Australia Ltd Buy 2,632 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	204	25	Ordinary
UBS Securities Australia Ltd Buy 15,126 1,561 UBS Securities Australia Ltd Buy 1,996 206 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 242 25 UBS Securities Australia Ltd Buy 6473 668 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 2,229 230 UBS Securities Australia Ltd Buy 2,229 230 UBS Securities Australia Ltd Buy 2,554 272 UBS Securities Australia Ltd Buy 2,554 272 UBS Securities Australia Ltd Buy 2,559 5,938 UBS Securities Australia Ltd Buy 2,654 760 UBS Securities Australia Ltd Buy 5,132	UBS Securities Australia Ltd Buy 15,126 UBS Securities Australia Ltd Buy 1,266 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,081 UBS Securities Australia Ltd Buy 2,02 UBS Securities Australia Ltd Buy 2,05 UBS Securities Australia Ltd Buy 2,05 UBS Securities Australia Ltd Buy 6,628 UBS Securities Australia Ltd Buy 6,628 UBS Securities Australia Ltd Buy 2,054 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 5,736 UBS Securities Australia Ltd Buy 5,736 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,342 UBS Securities Australia Ltd Buy	41-nu	UBS Securities Australia Ltd	Buv	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 9,048 UBS Securities Australia Ltd Buy 9,048 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,081 318 UBS Securities Australia Ltd Buy 242 24 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,1628 271 UBS Securities Australia Ltd Buy 2,229 230 UBS Securities Australia Ltd Buy 2,432 251 UBS Securities Australia Ltd Buy 2,432 261 UBS Securities Australia Ltd Buy 2,432 251 UBS Securities Australia Ltd Buy 2,432 253 UBS Securities Australia Ltd Buy 2,432 264	UBS Securities Australia Ltd Buy 9,912 UBS Securities Australia Ltd Buy 1,996 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,081 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 6,733 UBS Securities Australia Ltd Buy 5,729 UBS Securities Australia Ltd Buy 2,729 UBS Securities Australia Ltd Buy 2,739 UBS Securities Australia Ltd Buy 2,739 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy	In-14	II IRS Securities Australia I to	Bisy	15 126	1 561	Ordinan
UBS Securities Australia Ltd Buy 1,996 206 UBS Securities Australia Ltd Buy 3,159 3.26 UBS Securities Australia Ltd Buy 3,159 3.26 UBS Securities Australia Ltd Buy 640 66 UBS Securities Australia Ltd Buy 242 25 UBS Securities Australia Ltd Buy 67.2 27 UBS Securities Australia Ltd Buy 6.73 668 UBS Securities Australia Ltd Buy 6,1628 6.360 UBS Securities Australia Ltd Buy 5,153 5,938 UBS Securities Australia Ltd Buy 5,153 5,938 UBS Securities Australia Ltd Buy 5,153 664 UBS Securities Australia Ltd Buy 5,153 664 UBS Securities Australia Ltd Buy <	UBS Securities Australia Ltd Buy 1,996 UBS Securities Australia Ltd Buy 1,996 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,081 UBS Securities Australia Ltd Buy 2,620 UBS Securities Australia Ltd Buy 2,620 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy	17	A IBC Coourision Australia 1+d	66	0.001	000	Ordinary
UBS Securities Australia Ltd Buy 1,996 20b UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 640 66 UBS Securities Australia Ltd Buy 2,42 66 UBS Securities Australia Ltd Buy 6,473 66 UBS Securities Australia Ltd Buy 6,473 66 UBS Securities Australia Ltd Buy 6,473 68 UBS Securities Australia Ltd Buy 2,229 530 UBS Securities Australia Ltd Buy 2,229 530 UBS Securities Australia Ltd Buy 2,229 530 UBS Securities Australia Ltd Buy 2,229 593 UBS Securities Australia Ltd Buy 3,159 356 UBS Securities Australia Ltd Buy 3,159 36 UBS Securities Australia Ltd Buy 5,334 664 UBS Securities Australia Ltd Buy 6,132	UBS Securities Australia Ltd Buy 1,956 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 640 UBS Securities Australia Ltd Buy 2,626 UBS Securities Australia Ltd Buy 8,73 UBS Securities Australia Ltd Buy 6,73 UBS Securities Australia Ltd Buy 6,628 UBS Securities Australia Ltd Buy 2,064 UBS Securities Australia Ltd Buy 2,064 UBS Securities Australia Ltd Buy 2,739 UBS Securities Australia Ltd Buy 2,739 UBS Securities Australia Ltd Buy 2,739 UBS Securities Australia Ltd Buy 3,153 UBS Securities Australia Ltd Buy 5,739 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy <t< td=""><td></td><td>Opp Securities Australia Ltd</td><td>euy 5</td><td>105,5</td><td>050,1</td><td>Ordinary</td></t<>		Opp Securities Australia Ltd	euy 5	105,5	050,1	Ordinary
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 3,081 316 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 6,728 25 UBS Securities Australia Ltd Buy 6,1528 6,360 UBS Securities Australia Ltd Buy 6,1628 6,360 UBS Securities Australia Ltd Buy 2,054 2,17 UBS Securities Australia Ltd Buy 2,632 2,53 UBS Securities Australia Ltd Buy 5,538 5,938 UBS Securities Australia Ltd Buy 5,7,539 5,938 UBS Securities Australia Ltd Buy 5,7,539 5,938 UBS Securities Australia Ltd Buy 5,7,539 5,938 UBS Securities Australia Ltd Buy 5,634 764 UBS Securities Australia Ltd Buy <td>UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,081 UBS Securities Australia Ltd Buy 2,622 UBS Securities Australia Ltd Buy 2,622 UBS Securities Australia Ltd Buy 2,623 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 5,628 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,543 UBS Securities Australia Ltd Buy</td> <td>4</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>966't</td> <td>206</td> <td>Ordinary</td>	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,081 UBS Securities Australia Ltd Buy 2,622 UBS Securities Australia Ltd Buy 2,622 UBS Securities Australia Ltd Buy 2,623 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 5,628 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,543 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	966't	206	Ordinary
UBS Securities Australia Ltd Buy 3,159 3,6 UBS Securities Australia Ltd Buy 5,081 3,081 UBS Securities Australia Ltd Buy 6,473 6,6 UBS Securities Australia Ltd Buy 8,72 30 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 2,229 230 UBS Securities Australia Ltd Buy 2,054 212 UBS Securities Australia Ltd Buy 2,054 212 UBS Securities Australia Ltd Buy 3,159 5,938 UBS Securities Australia Ltd Buy 7,539 6,434 7,60 UBS Securities Australia Ltd Buy 6,434 7,64 7,60 UBS Securities Aus	UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 3 081 UBS Securities Australia Ltd Buy 640 UBS Securities Australia Ltd Buy 2 226 UBS Securities Australia Ltd Buy 8 72 UBS Securities Australia Ltd Buy 3 159 UBS Securities Australia Ltd Buy 6 6.239 UBS Securities Australia Ltd Buy 2 0.64 UBS Securities Australia Ltd Buy 2 1.239 UBS Securities Australia Ltd Buy 2 1.239 UBS Securities Australia Ltd Buy 5 7.539 UBS Securities Australia Ltd Buy 5 7.539 UBS Securities Australia Ltd Buy 5 7.539 UBS Securities Australia Ltd Buy 6 1.92 UBS Securities Australia Ltd Buy 6 1.92 UBS Securities Australia Ltd Buy 6 5.92 UBS Securities Australia Ltd Buy 6 5.92 UBS Securities Australia Ltd Buy 6 5.92 UBS Securities Australia Ltd	n-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 3,081 318 UBS Securities Australia Ltd Buy 640 66 UBS Securities Australia Ltd Buy 242 25 UBS Securities Australia Ltd Buy 2626 271 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,1628 356 UBS Securities Australia Ltd Buy 2,029 230 UBS Securities Australia Ltd Buy 2,054 271 UBS Securities Australia Ltd Buy 2,739 5,938 UBS Securities Australia Ltd Buy 2,739 5,938 UBS Securities Australia Ltd Buy 5,7539 5,938 UBS Securities Australia Ltd Buy 5,7539 5,938 UBS Securities Australia Ltd Buy 5,7539 5,938 UBS Securities Australia Ltd Buy 6,192 664 UBS Securities Australia Ltd Buy 5,7539 5,938 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 3.081 UBS Securities Australia Ltd Buy 242 UBS Securities Australia Ltd Buy 2.626 UBS Securities Australia Ltd Buy 2.626 UBS Securities Australia Ltd Buy 6.733 UBS Securities Australia Ltd Buy 6.628 UBS Securities Australia Ltd Buy 2.029 UBS Securities Australia Ltd Buy 2.034 UBS Securities Australia Ltd Buy 2.432 UBS Securities Australia Ltd Buy 5.7539 UBS Securities Australia Ltd Buy 5.7539 UBS Securities Australia Ltd Buy 5.7539 UBS Securities Australia Ltd Buy 6.434 UBS Securities Australia Ltd Buy 6.713 UBS Securities Australia Ltd Buy 6.732 UBS Securities Australia Ltd Buy 6.434 UBS Securities Australia Ltd Buy 6.543 UBS Securities Australia Ltd Buy 6.548 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 640 66 UBS Securities Australia Ltd Buy 242 25 UBS Securities Australia Ltd Buy 2626 271 UBS Securities Australia Ltd Buy 6473 668 UBS Securities Australia Ltd Buy 61,628 636 UBS Securities Australia Ltd Buy 61,628 6360 UBS Securities Australia Ltd Buy 2,054 212 UBS Securities Australia Ltd Buy 2,054 212 UBS Securities Australia Ltd Buy 2,153 94 UBS Securities Australia Ltd Buy 5,539 6,938 UBS Securities Australia Ltd Buy 7,539 5,938 UBS Securities Australia Ltd Buy 7,539 6,93 UBS Securities Australia Ltd Buy 7,539 6,93 UBS Securities Australia Ltd Buy 7,364 7,60 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434	UBS Securities Australia Ltd Buy 640 UBS Securities Australia Ltd Buy 242 UBS Securities Australia Ltd Buy 2.626 UBS Securities Australia Ltd Buy 872 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 6,1,628 UBS Securities Australia Ltd Buy 2,239 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 5,7539 UBS Securities Australia Ltd Buy 5,7539 UBS Securities Australia Ltd Buy 5,7539 UBS Securities Australia Ltd Buy 5,754 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buv	3,081	318	Ordinary
USS Securities Australia Ltd Buy 242 25 UBS Securities Australia Ltd Buy 2626 271 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,1529 236 UBS Securities Australia Ltd Buy 2,1229 230 UBS Securities Australia Ltd Buy 2,654 212 UBS Securities Australia Ltd Buy 2,1329 230 UBS Securities Australia Ltd Buy 2,1539 343 UBS Securities Australia Ltd Buy 3,159 356 UBS Securities Australia Ltd Buy 5,159 5,938 UBS Securities Australia Ltd Buy 5,159 5,938 UBS Securities Australia Ltd Buy 6,192 664 UBS Securities Australia Ltd Buy 6,192 664 UBS Securities Australia Ltd Buy 6,344 664 UBS Securities Australia Ltd Buy 6,	UBS Securities Australia Ltd Buy 242 UBS Securities Australia Ltd Buy 2.656 UBS Securities Australia Ltd Buy 872 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 6,628 UBS Securities Australia Ltd Buy 2,054 UBS Securities Australia Ltd Buy 2,054 UBS Securities Australia Ltd Buy 2,159 UBS Securities Australia Ltd Buy 5,159 UBS Securities Australia Ltd Buy 5,159 UBS Securities Australia Ltd Buy 5,159 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy <t< td=""><td>10-14</td><td>I IBS Sequitities Australia I td</td><td>Bix</td><td>640</td><td>99</td><td>Ordinary</td></t<>	10-14	I IBS Sequitities Australia I td	Bix	640	99	Ordinary
UBS Securities Australia Ltd Buy 2,626 271 UBS Securities Australia Ltd Buy 872 90 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,1628 6,380 UBS Securities Australia Ltd Buy 6,1628 6,380 UBS Securities Australia Ltd Buy 2,054 2,12 UBS Securities Australia Ltd Buy 2,432 2,51 UBS Securities Australia Ltd Buy 5,733 5,938 UBS Securities Australia Ltd Buy 5,733 5,938 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 659 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,344 644 UBS Securities Australia Ltd Buy 6,344 644 UBS Securities Australia Ltd Buy <	UBS Securities Australia Ltd Buy 2.626 UBS Securities Australia Ltd Buy 2.626 UBS Securities Australia Ltd Buy 6.473 UBS Securities Australia Ltd Buy 6.473 UBS Securities Australia Ltd Buy 6.1,628 UBS Securities Australia Ltd Buy 2.054 UBS Securities Australia Ltd Buy 2,632 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 5,759 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy		I DC Coording A setalia 14d	600	CASC	3 6	Ordinary Ordinary
UBS Securities Australia Ltd Buy 6,725 971 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,622 3,36 UBS Securities Australia Ltd Buy 2,229 230 UBS Securities Australia Ltd Buy 2,054 212 UBS Securities Australia Ltd Buy 3,159 5,938 UBS Securities Australia Ltd Buy 3,159 5,938 UBS Securities Australia Ltd Buy 7,539 5,938 UBS Securities Australia Ltd Buy 7,364 7,00 UBS Securities Australia Ltd Buy 6,192 654 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 654 UBS Securities Australia Ltd Buy 6,434 654 UBS Securities Australia Ltd Buy <t< td=""><td>UBS Securities Australia Ltd Buy 2.02.0 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 6,159 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,064 UBS Securities Australia Ltd Buy 2,159 UBS Securities Australia Ltd Buy 2,153 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 5,122 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 8,948 UBS Securities Australia Ltd Buy</td><td></td><td>Upo Commiss Australia Liu</td><td>, and</td><td>747</td><td>67</td><td>Ordinary</td></t<>	UBS Securities Australia Ltd Buy 2.02.0 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 6,159 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,064 UBS Securities Australia Ltd Buy 2,159 UBS Securities Australia Ltd Buy 2,153 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 5,122 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 8,948 UBS Securities Australia Ltd Buy		Upo Commiss Australia Liu	, and	747	67	Ordinary
UBS Securities Australia Ltd Buy 872 90 UBS Securities Australia Ltd Buy 6,473 668 UBS Securities Australia Ltd Buy 6,628 326 UBS Securities Australia Ltd Buy 2,029 230 UBS Securities Australia Ltd Buy 2,654 212 UBS Securities Australia Ltd Buy 2,632 251 UBS Securities Australia Ltd Buy 57,539 5,938 UBS Securities Australia Ltd Buy 57,539 5,938 UBS Securities Australia Ltd Buy 5,139 366 UBS Securities Australia Ltd Buy 5,139 6,192 6,192 UBS Securities Australia Ltd Buy 6,192 6,39 6,434 664 UBS Securities Australia Ltd Buy 6,434 664 109 UBS Securities Australia Ltd Buy 6,434 664 109 UBS Securities Australia Ltd Buy 6,434 664 109 UBS Securities Australia Ltd Buy	UBS Securities Australia Ltd Buy 6 473 UBS Securities Australia Ltd Buy 6 473 UBS Securities Australia Ltd Buy 5,159 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,634 UBS Securities Australia Ltd Buy 2,632 UBS Securities Australia Ltd Buy 5,159 UBS Securities Australia Ltd Buy 5,159 UBS Securities Australia Ltd Buy 5,159 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,643 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948	t :	Obs securities Australia Ltd	Buy	979'7	1/7	Ordinary
UBS Securities Australia Ltd Buy 6.473 668 UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 6,1628 6,360 UBS Securities Australia Ltd Buy 2,229 230 UBS Securities Australia Ltd Buy 2,054 212 UBS Securities Australia Ltd Buy 911 94 UBS Securities Australia Ltd Buy 5,539 5,938 UBS Securities Australia Ltd Buy 7,544 760 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 653 UBS Securities Australia Ltd Buy 6,192 654 UBS Securities Australia Ltd Buy 6,344 717 UBS Securities Australia Ltd Buy 6,344 664 UBS Securities Australia Ltd Buy 6,344 717 UBS Securities Australia Ltd Buy 6,344 717 UBS Securities Australia Ltd Buy 2,85	UBS Securities Australia Ltd Buy 6,473 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 61,628 UBS Securities Australia Ltd Buy 2,024 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 5,132 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 8,202 UBS Securities Australia Ltd Buy 8,202 UBS Securities Australia Ltd Buy 8,202 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	872	8	Ordinary
UBS Securities Australia Ltd Buy 3159 326 UBS Securities Australia Ltd Buy 61,628 6,360 UBS Securities Australia Ltd Buy 2,729 230 UBS Securities Australia Ltd Buy 2,654 212 UBS Securities Australia Ltd Buy 2,753 94 UBS Securities Australia Ltd Buy 57,530 5,938 UBS Securities Australia Ltd Buy 5,159 366 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,132 654 UBS Securities Australia Ltd Buy 6,132 654 UBS Securities Australia Ltd Buy 6,132 654 UBS Securities Australia Ltd Buy 6,344 664 UBS Securities Australia Ltd Buy 6,344 664 UBS Securities Australia Ltd Buy 6,344 664 UBS Securities Australia Ltd Buy 6,346 7,17 UBS Securities Australia Ltd Buy 6	UBS Securities Australia Ltd Buy 3.159 UBS Securities Australia Ltd Buy 61,628 UBS Securities Australia Ltd Buy 2,024 UBS Securities Australia Ltd Buy 2,054 UBS Securities Australia Ltd Buy 2,132 UBS Securities Australia Ltd Buy 91,1 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 5,7,534 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 8,948	un-14	UBS Securities Australia Ltd	Buy	6,473	999	Ordinary
UBS Securities Australia Ltd Buy 61,628 6,360 UBS Securities Australia Ltd Buy 2,729 230 UBS Securities Australia Ltd Buy 2,432 251 UBS Securities Australia Ltd Buy 2,432 251 UBS Securities Australia Ltd Buy 5,739 593 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 659 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,344 644 UBS Securities Australia Ltd Buy 8,346 717 UBS Securities Australia Ltd Buy 8,359 2,951 UBS Securities Australia Ltd Buy 8,595 2,951 UBS Securities Australia Ltd Buy <td< td=""><td>UBS Securities Australia Ltd Buy 61,628 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,632 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 5,156 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,394 UBS Securities Australia Ltd Buy 6,394 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348</td><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,159</td><td>326</td><td>Ordinary</td></td<>	UBS Securities Australia Ltd Buy 61,628 UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,632 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 5,156 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,394 UBS Securities Australia Ltd Buy 6,394 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348	un-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
UBS Securities Australia Ltd Buy 2,229 230 UBS Securities Australia Ltd Buy 2,054 212 UBS Securities Australia Ltd Buy 2,432 251 UBS Securities Australia Ltd Buy 57,539 5,938 UBS Securities Australia Ltd Buy 3,159 5,938 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 653 UBS Securities Australia Ltd Buy 6,192 654 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 654 UBS Securities Australia Ltd Buy 6,434 654 UBS Securities Australia Ltd Buy 6,948 717 UBS Securities Australia Ltd Buy 6,948 717 UBS Securities Australia Ltd Buy 6,948 717 UBS Securities Australia Ltd Buy 2,951 175 UBS Securities Australia Ltd Buy 2	UBS Securities Australia Ltd Buy 2,229 UBS Securities Australia Ltd Buy 2,054 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 91,1 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 5,195 UBS Securities Australia Ltd Buy 6,132 UBS Securities Australia Ltd Buy 6,132 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 8,948 UBS Securities Australia Ltd Buy 8,948	un-14	UBS Securities Australia Ltd	Sa.	61.628	6.360	Ordinary
UBS Securities Australia Ltd Buy 2,054 272 UBS Securities Australia Ltd Buy 2,054 271 UBS Securities Australia Ltd Buy 2,432 251 UBS Securities Australia Ltd Buy 57,539 5,938 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 659 UBS Securities Australia Ltd Buy 6,192 659 UBS Securities Australia Ltd Buy 6,344 644 UBS Securities Australia Ltd Buy 6,344 644 UBS Securities Australia Ltd Buy 6,344 671 UBS Securities Australia Ltd Buy 6,346 717 UBS Securities Australia Ltd Buy 8,345 717 UBS Securities Australia Ltd Buy 8,3595 2,951 UBS Securities Australia Ltd Buy 1,12,90 1,156	UBS Securities Australia Ltd Buy 2,054 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 2,432 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 5,7539 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,634 UBS Securities Australia Ltd Buy 6,643 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948	n-14	LIBS Securities Australia Ind	Biv	9266	230	Ordinary
UBS Securities Australia Ltd Buy 2,423 212 UBS Securities Australia Ltd Buy 2,433 251 UBS Securities Australia Ltd Buy 91 94 UBS Securities Australia Ltd Buy 5,539 5,938 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 644 UBS Securities Australia Ltd Buy 6,948 717 UBS Securities Australia Ltd Buy 7,855 7,951	UBS Securities Australia Ltd Buy 2,034 UBS Securities Australia Ltd Buy 2,439 UBS Securities Australia Ltd Buy 911 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,262 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348	,	1100 Countition Australia Ltd	in a	2,223	25.5	Countries
UBS Securities Australia Ltd Buy 4,432 251 UBS Securities Australia Ltd Buy 911 94 UBS Securities Australia Ltd Buy 57,539 5,938 UBS Securities Australia Ltd Buy 7,544 766 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 6,348 717 UBS Securities Australia Ltd Buy 6,348<	UBS Securities Australia Ltd Buy 91.43.2 UBS Securities Australia Ltd Buy 91.1 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 7,364 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 8,948	:	Obsideration Advisor and Lind	had	2,034	717	Ordinary
UBS Securities Australia Ltd Buy 57,539 94 UBS Securities Australia Ltd Buy 57,539 5,938 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,434 643 UBS Securities Australia Ltd Buy 5,262 543 UBS Securities Australia Ltd Buy 2,562 543 UBS Securities Australia Ltd Buy 2,555 2,948 717 UBS Securities Australia Ltd Buy 2,555 2,948 717 UBS Securities Australia Ltd Buy 2,555 2,941 UBS Securities Australia Ltd Buy 2,555 2,951	UBS Securities Australia Ltd Buy 911 UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 3,153 UBS Securities Australia Ltd Buy 7,364 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 5,262 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348	4-14	UBS Securities Australia Ltd	Buy	2,432	751	Ordinary
UBS Securities Australia Ltd Buy 5,539 5,938 UBS Securities Australia Ltd Buy 3,159 3,59 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 5,262 543 UBS Securities Australia Ltd Buy 6,948 7,17 UBS Securities Australia Ltd Buy 6,948 7,17 UBS Securities Australia Ltd Buy 6,544 7,17 UBS Securities Australia Ltd Buy 6,548 7,17 UBS Securities Australia Ltd Buy 1,200 1,165	UBS Securities Australia Ltd Buy 57,539 UBS Securities Australia Ltd Buy 3,152 UBS Securities Australia Ltd Buy 7,364 UBS Securities Australia Ltd Buy 6,132 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348 UBS Securities Australia Ltd Buy 6,348	un-14	UBS Securities Australia Ltd	Buy	911	94	Ordinary
UBS Securities Australia Ltd Buy 3,159 326 UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 5,262 543 UBS Securities Australia Ltd Buy 6,948 717 UBS Securities Australia Ltd Buy 28,595 2,551 IRS Securities Australia Ltd Buy 11,200 1,165	UBS Securities Australia Ltd Buy 3,159 UBS Securities Australia Ltd Buy 7,364 UBS Securities Australia Ltd Buy 6,132 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 6,848	un-14	UBS Securities Australia Ltd	Виу	57,539	5,938	Ordinary
UBS Securities Australia Ltd Buy 7,364 760 UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 5,262 543 UBS Securities Australia Ltd Buy 26,595 2,948 UBS Securities Australia Ltd Buy 26,595 2,951 UBS Securities Australia Ltd Buy 1,200 1,165	UBS Securities Australia Ltd Buy 7,364 UBS Securities Australia Ltd Buy 6,192 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 5,262 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 28,595	un-14	UBS Securities Australia Ltd	Buv	3.159	326	Ordinary
UBS Securities Australia Ltd Buy 6,192 639 UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 5,262 543 UBS Securities Australia Ltd Buy 6,948 717 UBS Securities Australia Ltd Buy 28,595 2,951 IRS Securities Australia Ltd Buy 13,990 1,166	UBS Securities Australia Ltd Buy 6,132 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 6,434 UBS Securities Australia Ltd Buy 5,262 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 28,595	14	UBS Securities Australia I to	Biv	7 364	760	Ordinary
UBS Securities Australia Ltd Buy 6,132 6,34 664 UBS Securities Australia Ltd Buy 5,262 543 664 UBS Securities Australia Ltd Buy 5,262 543 717 UBS Securities Australia Ltd Buy 28,595 2,551 IRS Securities Australia Ltd Buy 28,595 2,551 IRS Securities Australia Ltd Ruy 11,200 1,165	UBS Securities Australia Ltd Buy 6,124	1	LIBC Con intion Australia 144	9.00	6 100	OC.	Carling
UBS Securities Australia Ltd Buy 6,434 664 UBS Securities Australia Ltd Buy 5,262 5,43 UBS Securities Australia Ltd Buy 6,948 7,17 UBS Securities Australia Ltd Buy 6,948 7,17 UBS Securities Australia Ltd Buy 6,549 2,651 IRS Securities Australia Ltd Buy 1,200 1,165	UBS Securities Australia Ltd Buy 6,434 UGBS Securities Australia Ltd Buy 5,262 UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 28,595	:	Obsiseculities Australia Ltd	eny	0,132	600	Oranary
UBS Securities Australia Ltd Buy 5,262 543 UBS Securities Australia Ltd Buy 6,948 717 UBS Securities Australia Ltd Buy 28,595 2,951 IRS Securities Australia Ltd Ruy 11,290 1,165	UBS Securities Australia Ltd Buy 5,262 UBS Securities Australia Ltd Buy 6,948 UUSS Securities Australia Ltd Buy 28,595	1n-14	UBS Securities Australia Ltd	Buy	6,434	664	Ordinary
UBS Securities Australia Ltd Buy 6.948 717 UBS Securities Australia Ltd Buy 28,595 2,951 IRS Securities Australia Ltd Buy 11,290 1,165	UBS Securities Australia Ltd Buy 6,948 UBS Securities Australia Ltd Buy 28,595	nr-14	UBS Securities Australia Ltd	Buy	5,262	243	Ordinary
UBS Securities Australia 12d Buy 28,595 2,951 IRS Securities distributed in the Researchies	UBS Securities Australia Ltd Buy 28,595	Jn-14	UBS Securities Australia Ltd	Buy	6,948	717	Ordinary
I IRS Securities Australia Ird Ruy 11.200 11.65		Jn-14	UBS Securities Australia Ltd	Buy	28,595	2,951	Ordinary
	I DC Consistent A second in test		I IDO Consisting Australia Lad		000 11	1,100	
UDS SECULITIES AUSTRALIA LTG DUY	UBS SECUTIVES AUSTRAIRA LTG BUY	1	חסט סברתו וובס אחסת שוש דות	buy	2571	1,1001	Crairiary
	'0'	1000	I Bo Secretary A retrains +0		202	52	Ordinan

1-Jun-14	1 100 C 12 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			***	
1-Jun-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,143	35.2	Ordinary
Jun-14	UBS Securities Australia Ltd	Buv	3.110	321	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	3,149	325	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	699'5	585	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Виу	3,159	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	16,415	1,694	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,735	179	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	3,149	325	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Виу	3,033	313	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	523	54	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	620	64	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,392	350	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Вцу	9,206	950	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	48	2	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	7,364	760	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	15,959	1,647	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,636	272	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	48	2	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
JUN-14	UBS Securities Australia Ltd	Buy	2,645	273	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	4,089	422	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	11,192	1,155	Ordinary
- 10D-14	UBS Securities Australia Ltd	Buy	9,639	375	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	19,410	2,000	Ordinary
100 -14	UBS Securities Australia Ltd	Buy	437	45	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	2,912	300	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	6,706	169	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	2,892	298	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,214	228	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	184	19	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	437	45	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,719	177	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	71,728	7,387	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	15,720	1,619	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,165	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	971	100	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	50,162	5,166	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	21,750	2,240	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	6,040	622	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,175	430	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	16,507	1,700	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,156	325	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	272	28	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	3.639	375	Ordinary
Jun-14	UBS Securities Australia Ltd	BIN	3 156	375	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	126	100	Ordinary
1-lim-14	LIBS Securities Australia Ltd	Bire	3 156	325	Ordinary
1-hm-14	LiBS Securities Australia 14d	Box	5,150	23	Cidinary
1 1 1	Liber Cognition Australia Ltd	, and	0316	ò	Ordinary
1	ODS Securities Australia Ltd	ána	001.0	525	Ordinary
101-1	ODS SECULINES AUSTRAINS LIQ	, and	91/16	283	Orginary
1 - I	UBS Securities Australia Ltd	Buy	985,5	349	Ordinary
1-JUN-14	UBS Securities Australia Ltd	Buy	612	63	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,408	145	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,262	233	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	670,1	106	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	476	49	Ordinary
-Un-14	UBS Securities Australia Ltd	Buy	291	30	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,165	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,418	146	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	39	4	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	670	C	Ordinan
				60	Clemany
-kın-14	UBS Securities Australia Ltd	Buv	12.930	1 333	Ordinary

11-4-0-17	I IBC Con critics Attended 140	, d	1626	324	, and a signar
1-lin-14	LIBS Securities Australia 11d	Riv	4,0,4	2,7	Ordinary
14	UBS Securities Australia Ltd	Buy	8,731	901	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,694	278	Ordinary
11-Jun-14	UBS Securities Australia Ltd	3rv	2.888	298	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	16,599	1,713	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	17,442	1,800	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,364	244	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	9,564	286	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	5,785	265	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	296'9	718	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,663	378	Ordinary
4	UBS Securities Australia Ltd	Buy	6,793	701	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	3,140	324	Ordinary
4	UBS Securities Australia Ltd	Buy	1,202	124	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,705	176	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,740	386	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,180	225	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	21,153	2,183	Ordinary
₽ ;	UBS Securities Australia Ltd	Buy	78,392	2,930	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	76,221	2,706	Ordinary
1-Jun-14	Upo Securities Australia Liu	eny	/0/	5/5	Orginary
1-Jun-14	UBS Securities Australia Ltd	env.	4,632	8/4	Ordinary
- Jun 14	Obs Securities Australia Ltd	śna.	00,300	003	Ordinary
1	BS Securities Australia 14d	9.50	6,200	767'	Ordinary
1-line14	I IRS Securities Australia 14d	2 8	0000	5	Ordinary
1-110-14	I IRS Securities Australia 14d	600	7 133	361	Ordinary
1-hin-14	I IRS Securities Australia 11d	Risk	261,	773	Ordinary
- h m-14	I IRS Securities Australia 11d	Buy	12 713	1312	Ordinan
1 4	LIBS Securities Australia Ind	Bird	17 442	1,312	Ordinary
1-Jun-14	UBS Securities Australia (1d	Silv	7.457	769	Ordinary
1-lun-14	ILIRS Securities Australia Ltd	Biv	77 304	7 987	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	10	-	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	685'9	089	Ordinary
4	UBS Securities Australia Ltd	Buy	1,463	151	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	13.605	1.404	Ordinary
1-Jun-14	UBS Securities Australia i td	Buv	5 775	295	Ordinary
1-Jun-14	IUBS Securities Australia Ltd	Buy	502.9	269	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	101	-	Ordinary
4	UBS Securities Australia Ltd	Buy	1,754	181	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	9'9	889	Ordinary
4	UBS Securities Australia Ltd	Buy	3,149	325	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	20,233	2,088	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	101	-	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	5.174	534	Ordinary
4	UBS Securities Australia Ltd	Buy	3,779	390	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	174	18	Ordinary
1-lun-14	UBS Securities Australia I td	Bliv	1 822	188	Ordinary
1-lin-14	LIBS Securities Australia Ltd	Bir	3,674	374	Ordinary
1-110-17	I IBC Counties Australia 14d	, Co. C	750,0	100	Ordinary
: :	I IDC Consisting Augustia Ltd	, , , ,	05/10	1717	Ordinary
1-3011-14	Obsider Australia Lid	Buy	00001	11/1/	Ordinary
4 2	UBS Securities Australia Ltd	Buy	4,535	468	Ordinary
1-3011-14	UDS Securities Australia Ltd	bad)	oco'l	60.	Ordinary
- 70:1-14	Obsideration Australia Lid	Buy	D 1	7	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy B. I.	10,756	011,1	Ordinary
<u>.</u> ;	UBS Securities Australia Ltd	Buy	3,5/5	505	Orginary
1-Jun-14	UBS Securities Australia Ltd	Buy	079's	280	Ordinary
4	UBS Securities Australia Ltd	Buy	959	66	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	8,353	862	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,017	105	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	4 167	000	
ŀ				430	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buv	152,230	15.710	Ordinary

UBS Securities Australia Ltd	Buy	3,153	325	Ordinary
UBS Securities Australia Ltd	Buy	1,756	181	Ordinary
UBS Securities Australia Ltd	Buy	3,162	326	Ordinary
curities Australia Ltd	Buy	3,162	326	Ordinary
UBS Securities Australia Ltd	Buy	2,357	243	Ordinary
UBS Securities Australia Ltd	Buy	3,153	325	Ordinary
UBS Securities Australia Ltd	Buy	11,427	1,178	Ordinary
curities Australia Ltd	Buv	2377	245	Ordinary
UBS Securities Australia Ltd	Buy	13 745	1 417	Ordinan
RS Securities Australia Ltd	200	9093	001	
	600	0.20,0	3	G G I I
Obs securities Australia Ltd	aoy	14,048	-,449	Ordinary
curities Australia Ltd	Buy	01	-	Ordinary
ecurities Australia Ltd	Buy	6,033	622	Ordinary
Sourities Australia Ltd	>iig	4 443	458	Ordinary
I IRC Constitute Australia Ltd	200	5 033	503	
DO Commission Automatic Land	bay.	0,000	770	Crainary
curines Australia Ltd	Buy	000,78	10,000	Ordinary
JBS Securities Australia Ltd	Buy	970	100	Ordinary
UBS Securities Australia Ltd	Buy	15,753	1,624	Ordinary
UBS Securities Australia Ltd	Buy	157,509	16,238	Ordinary
ecurities Australia Ltd	N A	R0 297	8778	Ordinan
ecurities Australia (td	Bary	4171	V3V	Crainson
IOC Consistent Assessment and	Č.	200000	200	(in in in
ecurities Australia Ltd	Buy	56,503	5,825	Ordinary
JBS Securities Australia Ltd	Buy	118,049	12,170	Ordinary
UBS Securities Australia Ltd	Buy	3,705	382	Ordinary
JBS Securities Australia Ltd	Buy	2002	702	Ordinary
An initiae Australia I to	D. C.	T C C	-	in line
IDC Contribut Assetsolis 14d	6.0	2000	- 6	, in a
ecolities Australia Ltd	, and	2,425	067	Ordinary
JBS Securities Australia Ltd	Buy	1,562	161	Ordinary
JBS Securities Australia Ltd	Buy	3,531	364	Ordinary
UBS Securities Australia Ltd	Buy	3,531	364	Ordinary
ecurities Australia Ltd	Buy	8,672	894	Ordinary
ecurities Australia Ltd	Buv	1.872	193	Ordinary
UBS Securities Australia Ltd	Buv	2.862	295	Ordinary
JBS Securities Australia Ltd	2.0	1 562	161	Ordinary
JBS Securities Australia Ltd	Bliv	1 358	140	Ordingo
185 Securities Australia Ltd	S S	06		Ordings
to mittee Americal and		20.00	1 93,	S S S S S S S S S S S S S S S S S S S
DDS Securities Australia Ltd	pony	4,040	403	Ordinary
ecurities Australia Ltd	Buy	67877	767	Ordinary
UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
JBS Securities Australia Ltd	Buy	358	37	Ordinary
UBS Securities Australia Ltd	Buy	494	51	Ordinary
UBS Securities Australia Ltd	Biv	2.471	255	Ordinary
Contribut American In Land	600	030	557	Quintally Ordinally
eculties Australia Ltd	Buy	358	3/	Ordinary
UBS Securities Australia Ltd	Buy	3,149	325	Ordinary
ecurities Australia Ltd	Buy	6,250	645	Ordinary
UBS Securities Australia Ltd	Buv	358	37	Ordinary
ecurities Australia I to	Birv	2,005	207	Vaccina
At Antrope Automotive Leaf	600	2 1 40	200	O COLUMN
ODS Securities Australia Ltd	ng.	0,149	270	Ordinary
ecurities Australia Ltd	Buy	659	80	Ordinary
JBS Securities Australia Ltd	Buy	37,791	3,900	Ordinary
UBS Securities Australia Ltd	Buy	3,169	327	Ordinary
UBS Securities Australia Ltd	81%	1 269	131	Ordinary
Sociarities Australia 14d	200	2 731	200	decipo
Committee A committee of the committee o	600	16116	Sign	O'C' III
Opp Securities Australia LIG	Áng	OSS.	507	Ordinary
ecurities Australia Ltd	Buy	4,167	430	Ordinary
UBS Securities Australia Ltd	Buy	1,415	146	Ordinary
UBS Securities Australia Ltd	Buy	1,454	150	Ordinary
UBS Securities Australia Ltd	Buy	4,932	605	Ordinary
UBS Securities Australia Ltd	Buy	533	55	Ordinary
Politities Australia 114	Bir	29.020	000 %	Codingo
The second of th	on)	0/0/62	000,0	Cidinaly
UBS Securities Australia Ltd	Buy	2,471	255	Ordinary
JBS Securities Australia Ltd	Buy	1,996	506	Ordinary
UBS Securities Australia Ltd	Bliv	2,149	325	Ordinary
Consistent Assessment Life	600	001.0	275	O Oil iai
Obs securities Australia Ltd	ens	- 7		
			/ς7	Orginary
UBS Securities Australia I to	Bile	4 438	757	Ordinary

Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinan
006	305,	1,468	597	414	1,018	1,146	735	208	894	784	282	326	119	926	282	297	893	1,170	6,390	325	5,710	263	3.830	895	3,421	411	230	069	108	5,39	4,609	430	4,608	5,392	295	4,842	4 495	5,505	4,496	430	5,504	4,358	1,560	4,082	4,010	26	293	766	7,932	1,631	611	2,445	2,422	3,000	4 140	728	1.141	1881	385	S
8,730	12,610	14,240	155,7	4,016	9,875	11,116	7,130	4,928	8,672	7,755	2,735	3,162	1,154	9,467	2,735	2,881	8,662	11,349	61,983	3,153	792,38/	2,5,5	37.151	8,682	33,184	3,987	2,231	6,693	1,048	52,293	44,707	4,171	44,698	52,302	2,862	40,967	43 602	53,399	43,611	4,171	53,389	42,273	15,132	39,595	33 771	897	2.842	7,430	76,940	15,821	5,927	23,717		001,62	7,507 40,158	7 062	11.068	8,643	9,555	17563
8uy	, and	Buy	Buy	Buy	Buy	Buy	Buy	Ang	Buy	gnà	Buy	8uy	Buy	Ruy B. S.	Buy	Bir	Buy	and a	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	buy b. a.	Ris Sur	RIN DO	Bey	Buy	Buy	(5)																											
UBS Securities Australia Ltd	DES Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia 1td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	DBS Securities Australia Ltd	UBS Securities Australia i td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	IRS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	the state of the s
11-Jun-14	Ť	1	Ť	11-Jun-14	1	T	T	Ť	11-Jun-14	T	T	- 1		11-Jun-14	- 1	١		- 1	11-Jun-14	Ť	11-Jun-14	Ť	Ť	Ι-	Ē	Ī		T	-	11 11 11 14	ı		П	T	11-Jun-14	T	Τ	1	1	11-Jun-14			-1	-	11.1.1.1	1	1	Г	Г	11-Jun-14		П	T	Ť	11-110-14	T	Τ	1	l	ı

	The restance of the control of	22	1,0	ō	Ordinary
1-Jun-14	UBS Securities Australia Ltd	ío e	/8	,	or in the
1-lun-14	LIRS Securities Australia 3 td	Buy	1/8/7	1 250	Ordinary
1-lun-14	I IRS Securities Australia 11d	Bany	C/1'C1	055	Ordinary
1 1 10 14	UBS Securities Australia Ltd	Buy	2,638	7/7	Ordinary
1	I IBC Countries Australia Lea	buy b	16	2 5	Ordinary
1-lin-14	LBS Securities Australia Ltd	Bay	39	400	Ordinary
2	I IRS Securities Australia I to	600	260 6	010	Ordinary
- lin-14	BS Societies Australia Ltd	6.0	2,00	217	Ocurrany
1-hin-14	UBS Securities Australia Ltd	Bes	27.0	36	Ordinary Statistics
1-lun-14	I IRS Securities Australia I to	, a	803 C	27.0	Ordinary
1-lun-14	UBS Securities Australia 11d	Big	3,518	372	Ordinary
1	LIBS Securities Australia 11d	(a)	245	2/2	Ordinary
1-lim-14	I IRS Securities Australia 11d	(a)	242	220	Granary
1-1-1	I IRC Committee Americal and	one d	216,2	1 200	Crainary
1 1 1 1 1	LIBS Countries Australia Ltd	Sug.	12,558	ans'1	Ordinary
1,11	LIBS Countities Australia Ltd	buy	67179	838	Ordinary
1-in-14	J. IRC Securities Australia 144	, and	12 600	76	Ordinary
114	(IBS Securities Australia 11d	Bus	12,800	1,004	Ordinary
1-1 in-14	I IRS Sequirities Australia 11d	and a	210,01	248	Ordinary
1-lin-14	I IRS Sequitities Australia 11d	Bar	2,400	240	Ordinary
1 m 14	I IBC Counties Australia I+d	903	67	2	Ordinary
1 1	I IBC Counties Australia 14d	spa)	600'1	40.	Orainary
1 1	1 IDS Constitutes Australia Ltd	e e	67	0 1	Orginary
4	I IBS Securities Australia 11d	B. E.	104	n (Ordinary
-Jun-14	UBS Sequities Australia (1d	B. S.	1 086	112	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Bis	771	91	ordina o
1-Jun-14	UBS Securities Australia Ltd	Bliv	200	2	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Виу	20	-	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	2.803	289	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	29		Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	89	7	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	378	39	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	629	70	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	20	r	Ordinary
1-Jun-14	UBS Securities Australia I td	(S)	22	0	Ordinary
1-lun-14	UBS Securities Australia I to	() (a)	165	0 1	Ordinary
1-110-14	I IBS Sociution Australia 14d	00.00	2	-	Coloniary
1-110-14	I IBS Securities Australia 14d	in a	2 6	- 6	Ordinary
1-hm-14	LIRS Securities Australia 14d	, a	1 883	200	Ordinary
1	Under Committee Australia Lid	Suy B	7887	194	Ordinary
11-4-0-14	I IRC Countries Australia 14d	knga	2,629	1/7	Orginary
114	Libos Countries Australia Ltd	Buy	4,305	450	Ordinary
#1-JUL-1	UBS Securities Australia Ltd	Buy	67	m	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,027	509	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	766	79	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,232	127	Ordinary
4-1	UBS Securities Australia Ltd	Buy	10	-	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	29	Э	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	59	3	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	708	73	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	640	99	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	59	3	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	29	Э	Ordinary
4	UBS Securities Australia Ltd	Buy	10	•	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	1,630	168	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	6,518	672	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	97	10	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	6,392	629	Ordinary
1-inn-14					1
1	IUBS Securities Australia Ltd	Buv	3.153	325	Ordinary

11-Jun-14 (11-Jun-14 (UBS Securities Australia Ltd	Buv	3.527	364	Crains
	1 IBC Contribute Australia Ltd		-1212	. 5.5	Crumary
\Box	ODS SECURIES AUSUAINA LIQ	Buy	5,126	529	Ordinary
71	UBS Securities Australia Ltd	Buy	911	8	Ordinary
Т	UBS Securities Australia Ltd	Buy	3,513	362	Ordinary
	UBS Securities Australia Ltd	eny	14,558	005,1	Ordinary
T	UBS contribes Australia Ltd	any D. J.	2,039	2/5	Orginary
T	UBS Securities Australia Ltd	Buy	700,0 700,0	070	Ordinary
	UBS Securities Australia Ltd	Bcv	4.171	430	Ordinary
1	UBS Securities Australia Ltd	Buy	3,156	325	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	13,570	1,399	Ordinary
	UBS Securities Australia Ltd	Buy	10,720	1,104	Ordinary
Ī	UBS Securities Australia Ltd	Buy	3,321	342	Ordinary
- 1	UBS Securities Australia Ltd	Buy	1,911	197	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
- 1	UBS Securities Australia Ltd	Buy	1,077		Ordinary
1-Jun-14	IBS Societies Australia Ltd	èng.	7/5	2 00	Ordinary
	LIRS Securities Australia Ind	Rin	16 151	1 665	Ordinary
		Blay	1874	188	Ordinary
	JBS Securities Australia Ltd	Buy	175	31	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	2.903	565	Ordinary
1		Buv	0.030	026	Ordinary
		Buy	3,156	325	Ordinary
		Buy	3.156	325	Ordinary
ı		Buy	4,855	200	Ordinary
П	UBS Securities Australia Ltd	Buy	388	40	Ordinary
٦		Buy	3,156	325	Ordinary
_	Securities Australia Ltd	Buy	3,156	325	Ordinary
Ī		Buy	066	102	Ordinary
٦		Buy	31,606	3,255	Ordinary
7		Buy	7,059	727	Ordinary
Ť	UBS Securities Australia Ltd	Buy	3,156	325	Ordinary
		Buy	24,265	2,499	Ordinary
Ī		Buy	3,156	325	Ordinary
T		Buy	8,059	830	Ordinary
	Securities Australia Ltd	Buy	2,039	210	Ordinary
1-Jun-14		Buy	3,156	325	Ordinary
T	UBS Securities Australia Ltd	Buy	33,985	3,500	Ordinary
Ť		Buy	5,389	555	Ordinary
1-101-14		bug.	2,035	292	Ordinary
Ť		Bin,	3 156	375	Ordinary
T		Birv	252	25.0	Ordinary
T	UBS Securities Australia Ltd	Buy	9.671	266	Ordinary
11-Jun-14		Buv	8.643	168	Ordinary
Т		Buy	45,377	4,678	Ordinary
Γ	UBS Securities Australia Ltd	Buy	19,449	2,005	Ordinary
Г		Buy	3,298	340	Ordinary
	UBS Securities Australia Ltd	Buy	59	8	Ordinary
1-Jun-14		Buy	11,097	1,144	Ordinary
	Securities Australia Ltd	Buy	7,324	755	Ordinary
	Securities Australia Ltd	Buy	3,929	405	Ordinary
1	Securities Australia Ltd	Buy	29	3	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	59	m	Ordinary
ł		Buy	6,324	652	Ordinary
- [UBS Securities Australia Ltd	Buy	524	72	Ordinary
ı		Buy	427	44	Ordinary
ı		Buy	165	17	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	29	m	Ordinary
T		Buy	29	m	Ordinary
T	UBS Securities Australia Ltd	Buy	66	4	Ordinary
1-Jun-14		Buy	15,743	1,623	Ordinary
Т		Buy	8,410	ýg.	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	16,248	1,675	Ordinary

650 Securine Australia LID BM BAS BS 652 650 Securine Australia LID BM 8,458 8,73 650 Securine Australia LID BM 8,458 8,73 650 Securine Australia LID BM 8,458 8,75 650 Securine Australia LID BM 8,658 8,75 650 Securine Australia LID BM 8,75 7,75 650 Securine Australia LID BM 1,550 1,430 650 Securine Australia LID BM 1,550 1,430 650 Securine Australia LID BM 1,135 1,17 650 Securine Australia LID BM 1,12 1,17	1					Cramary
Big Securities Anatorials Led Buy 6,458 6,458 1,55	l.n.14	I IRS Securities Australia 14d	Britis	19141	000	
March March March May 3753 375	Jun-14	UBS Securities Australia Ltd	Buv	8 458	27.	Ordinary Andinary
March March May	-Jun-14	UBS Securities Australia Ltd	Buy	3 153	305	Spino C
USS Securities Anabraila Ltd Buy 6,956 9,10 <	lin-14	1 IRS Securities Australia I to	Buck	000	25.0	500
Big Securities Australia Ltd Buy 5,956 47.7 UBS Securities Australia Ltd Buy 7,956 47.9 UBS Securities Australia Ltd Buy 7,956 47.9 UBS Securities Australia Ltd Buy 7,956 7.9 UBS Securities Australia Ltd Buy 1,552 1.20 UBS Securities Australia Ltd Buy 1,552 1.20 UBS Securities Australia Ltd Buy 1,552 1.20 UBS Securities Australia Ltd Buy 1,562 1.20 UBS Securities Australia Ltd Buy 1,562 1.70 UBS Securities Australia Ltd Buy 2,132 2.9 UBS Securities Australia Ltd Buy 2,132 2.9 UBS Securities Australia Ltd Buy 2,132 2.0 UBS Securities Australia Ltd Buy 2,132 1.0 UBS Securities Australia Ltd Buy 2,132 2.0 UBS Securities Australia Ltd Buy 2,132 2.0 UBS Securities Australia Ltd Buy <t< td=""><td>110.17</td><td>LIBC Securities Australia 14d</td><td></td><td>0/000</td><td>٥</td><td>Ordinaly</td></t<>	110.17	LIBC Securities Australia 14d		0/000	٥	Ordinaly
USS Seconties Australia Ltd Bay 6.29 6.79 Loss Seconties Australia Ltd Bay 6.29 6.70 Loss Seconties Australia Ltd Bay 6.29 7.265 7.29 Loss Seconties Australia Ltd Bay 6.59 7.265 7.29 Loss Seconties Australia Ltd Bay 4.65 2.542 1.00 Loss Seconties Australia Ltd Bay 4.17 4.30 Loss Seconties Australia Ltd Bay 1.24 1.17 Loss Seconties Australia Ltd Bay 1.24 1.24 Loss Seconties Australia Ltd Bay 1.24 1.24 Loss Seconties Australia Ltd Bay 1.26 1.27 Loss Seconties Australia Ltd Bay 1.28 1.73 Loss Seconties Australia Ltd Bay 1.28 1.73 Loss Seconties Australia Ltd Bay 1.29 1.24 Loss Seconties Australia Ltd Bay 1.29 1.26 Loss Seconties Australia Ltd Bay 2.56 2.56 Loss	1 m-14	195 Securities Australia 144	600	500 0	210	O Command
URS Securities Australia Ltd Buy 7,265 7,49 10,10 URS Securities Australia Ltd Buy 1,455 7,29 1,51 URS Securities Australia Ltd Buy 1,455 1,52 1,52 URS Securities Australia Ltd Buy 2,667 2,542 1,51 URS Securities Australia Ltd Buy 2,667 2,542 1,51 URS Securities Australia Ltd Buy 2,667 1,52 1,52 URS Securities Australia Ltd Buy 2,73 2,67 2,68 URS Securities Australia Ltd Buy 1,135 1,17 1,40 URS Securities Australia Ltd Buy 2,100 2,100 1,40 URS Securities Australia Ltd Buy 1,58 1,72 1,40 URS Securities Australia Ltd Buy 1,58 1,70 1,40 URS Securities Australia Ltd Buy 1,58 1,70 1,40 URS Securities Australia Ltd Buy 1,58 1,70 1,40 URS Securities Australia Ltd	1 4	100 Coomition A continue Lea	900	066,0	412	Ordinary
UBS Securities Australia Ltd Buy 6,29 7,29 7,29 UBS Securities Australia Ltd Buy 4,52 1,60 UBS Securities Australia Ltd Buy 4,177 4,39 UBS Securities Australia Ltd Buy 2,657 1,60 UBS Securities Australia Ltd Buy 1,242 1,28 UBS Securities Australia Ltd Buy 1,242 1,28 UBS Securities Australia Ltd Buy 2,173 4,20 UBS Securities Australia Ltd Buy 2,173 2,22 UBS Securities Australia Ltd Buy 2,173 2,26 UBS Securities Australia Ltd Buy 2,100 1,20 UBS Securities Australia Ltd Buy 2,100 1,20 UBS Securities Australia Ltd Buy 2,100 2,100 UBS Securities Australia Ltd Buy 2,265 1,40 UBS Securities Australia Ltd Buy 2,265 1,40 UBS Securities Australia Ltd Buy 2,265 1,60 UBS Securities Australia Ltd		Tipe Committee Australia Ltd	ord	0,433	0/9	Ordinary
URS Securities Australia Ltd Buy 1,552 7,17 4,19 1,524 7,21 1,10	100	Coo Securities Australia Liu	Bucy	597'/	/49	Ordinary
UBS Securities Autarials Ltd Bluy 4,1552 160 UBS Securities Autarials Ltd Bluy 4,171 2,657 1,520 UBS Securities Autarials Ltd Bluy 4,171 2,527 1,430 UBS Securities Autarials Ltd Bluy 4,171 2,527 1,230 UBS Securities Autarials Ltd Bluy 4,171 2,237 1,281 UBS Securities Autarials Ltd Bluy 4,171 2,237 1,281 UBS Securities Autarials Ltd Bluy 4,171 2,237 1,281 UBS Securities Autarials Ltd Bluy 2,137 3,281 1,482 UBS Securities Autarials Ltd Bluy 2,137 3,281 1,482 UBS Securities Autarials Ltd Bluy 2,137 1,483 1,483 UBS Securities Autarials Ltd Bluy 2,137 1,522 1,401 UBS Securities Autarials Ltd Bluy 2,245 1,401 1,522 1,401 UBS Securities Autarials Ltd Bluy 2,245 1,429 1,429 1,429	Jun-14	UBS Securities Australia Ltd	Buy	6,994	721	Ordinary
UBS Securities Australia Ltd Buy 4,171 4,59 UBS Securities Australia Ltd Buy 2,652 1,59 UBS Securities Australia Ltd Buy 2,733 1,29 UBS Securities Australia Ltd Buy 2,733 1,29 UBS Securities Australia Ltd Buy 2,734 1,29 UBS Securities Australia Ltd Buy 2,735 1,73 UBS Securities Australia Ltd Buy 2,736 1,73 UBS Securities Australia Ltd Buy 2,935 1,73 UBS Securities Australia Ltd Buy 2,136 1,73 UBS Securities Australia Ltd Buy 2,122 2,56 UBS Securities Australia Ltd Buy 2,123 3,24 UBS Securities Australia Ltd Buy	- In	UBS Securities Australia Ltd	Buy	1,552	160	Ordinary
UBS Securities Australia Ltd Buy 24,657 2,542 UBS Securities Australia Ltd Buy 1,439 2,735 1,439 UBS Securities Australia Ltd Buy 2,735 2,82 1,439 1,430 1,439 1,430 1,439 1,444 1,439 1,444 1,439 1,444 1,439 1,444 1,439 1,444 1,439 1,444 1,439 1,444 1,439 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444 1,444	Jun-14	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
UBS Securities Australia Ltd Buy 1,455 150 UBS Securities Australia Ltd Buy 1,457 1,439 UBS Securities Australia Ltd Buy 2,735 1,23 UBS Securities Australia Ltd Buy 2,737 1,73 UBS Securities Australia Ltd Buy 2,93 3 UBS Securities Australia Ltd Buy 2,93 3 UBS Securities Australia Ltd Buy 2,93 3 UBS Securities Australia Ltd Buy 2,93 3,06 UBS Securities Australia Ltd Buy 2,93 3,07 UBS Securities Australia Ltd Buy 2,93 3,06 UBS Securities Australia Ltd Buy 2,93 3,07 UBS Securities Australia Ltd Buy 2,93 3,08 UBS Securities Australia Ltd Buy 1,32	Jun-14	UBS Securities Australia Ltd	Buy	24,657	2,542	Ordinary
UBS Securities Australia Ltd Buy 13.958 1,439 UBS Securities Australia Ltd Buy 2,173 282 UBS Securities Australia Ltd Buy 4,174 1,28 UBS Securities Australia Ltd Buy 2,175 1,28 UBS Securities Australia Ltd Buy 2,29 1,0 UBS Securities Australia Ltd Buy 2,9 2,0 UBS Securities Australia Ltd Buy 2,0 2	Jun-14	UBS Securities Australia Ltd	Buy	1,455	150	Ordinary
UBS Securities Australia Ltd Bby 2,735 282 UBS Securities Australia Ltd Bby 1,247 4,37 4,30 UBS Securities Australia Ltd Bby 1,355 117 4,30 UBS Securities Australia Ltd Bby 2,29 3 UBS Securities Australia Ltd Bby 2,29 3 UBS Securities Australia Ltd Bby 2,29 3 UBS Securities Australia Ltd Bby 2,95 101 UBS Securities Australia Ltd Bby 2,95 177 UBS Securities Australia Ltd Bby 2,95 3,05 UBS Securities Australia Ltd Bby 3,23 3,05 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buv	13.958	1 439	Ordinary
UBS Securities Australia Ltd Bby 4,171 439 UBS Securities Australia Ltd Bby 1,242 128 UBS Securities Australia Ltd Bby 29 13 UBS Securities Australia Ltd Bby 29 10 UBS Securities Australia Ltd Bby 29 10 UBS Securities Australia Ltd Bby 29 10 UBS Securities Australia Ltd Bby 2,925 101 UBS Securities Australia Ltd Bby 2,192 2,100 UBS Securities Australia Ltd Bby 2,192 2,501 UBS Securities Australia Ltd Bby 2,192 2,701 UBS Securities Australia Ltd Bby 1,292 2,501 UBS Securities Australia Ltd Bby 1,292 2,502 UBS Securities Australia Ltd Bby 1,292<	Jun-14	USS Securities Australia I to	Ni.	2 735	282	inchip
UBS Securities Australia Ltd BMy 1,242 4,25 UBS Securities Australia Ltd BMy 1,242 1,28 UBS Securities Australia Ltd BMy 29 3 UBS Securities Australia Ltd BMy 29 3 UBS Securities Australia Ltd BMy 29 3 UBS Securities Australia Ltd BMy 20,20 1,01 UBS Securities Australia Ltd BMy 2,102 2,26 UBS Securities Australia Ltd BMy 2,102 2,103 UBS Securities Australia Ltd BMy 6,108 2,103 UBS Securities Australia Ltd BMy 6,108 6,108 UBS Securities Australia Ltd BMy 6,108 6,108 UBS Securities Australia Ltd BMy 6,108 6,276 UBS Securities Australia Ltd BMy 1,231 1,052 UBS Securities Australia Ltd BMy 3,048 1,052 UBS Securities Australia Ltd BMy 3,048 1,052 UBS Securities Australia Ltd BMy 3	Jun-14	I IBS Securities Australia I td	Brit	1 171	707	Ordinary
UBS Securities Australia Ltd Buy 1,134 1,134 UBS Securities Australia Ltd Buy 1,134 1,134 UBS Securities Australia Ltd Buy 29 1,0 UBS Securities Australia Ltd Buy 29 1,0 UBS Securities Australia Ltd Buy 6,955 7,17 UBS Securities Australia Ltd Buy 2,192 2,50 UBS Securities Australia Ltd Buy 1,503 2,601 UBS Securities Australia Ltd Buy 1,193 2,707 UBS Securities Australia Ltd Buy 1,193 1,237 UBS Securities Australia Ltd Buy	11	LIDS Countries August 144	600	1/1/4	450	Oldanary
UBS Securities Australia Ltd Buy 1,135 117 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2,625 717 UBS Securities Australia Ltd Buy 2,1032 2,561 UBS Securities Australia Ltd Buy 2,1032 2,561 UBS Securities Australia Ltd Buy 2,1032 2,561 UBS Securities Australia Ltd Buy 2,1032 3,561 UBS Securities Australia Ltd Buy 2,1032 3,661 UBS Securities Australia Ltd Buy 2,1032 3,633 UBS Securities Australia Ltd Buy 2,265 7,74 UBS Securities Australia Ltd Buy 2,726 7,66 UBS Securities Australia Ltd Buy 2,76 4,67 UBS Securities Australia Ltd Buy 2,76 6,76 UBS Securities Australia Ltd Buy 2	1 1	ODS SECULINES AUSTRAND LIG	, and	747'	178	Ordinary
UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 10 UBS Securities Australia Ltd Buy 6,955 717 UBS Securities Australia Ltd Buy 6,955 717 UBS Securities Australia Ltd Buy 2,102 2,501 UBS Securities Australia Ltd Buy 5,132 2,501 UBS Securities Australia Ltd Buy 5,132 2,501 UBS Securities Australia Ltd Buy 5,132 2,501 UBS Securities Australia Ltd Buy 1,232 1,23 UBS Securities Australia Ltd Buy 1,232 1,23 UBS Securities Australia Ltd Buy 1,232 1,23 UBS Securities Australia Ltd Buy 1,23 1,23 UBS Securities Australia Ltd Buy 1,23 1,25 UBS Securities Australia Ltd Buy 2,135 1,25 UBS Securities Australia Ltd Buy 2,265 1,65 UBS Securities Australia Ltd Buy 2,	Jun-14	UBS Securities Australia Ltd	Buy	1,135	117	Ordinary
UBS Securities Australia Ltd Buy 97 10 UBS Securities Australia Ltd Buy 97 10 UBS Securities Australia Ltd Buy 6,955 717 UBS Securities Australia Ltd Buy 1,582 161 UBS Securities Australia Ltd Buy 2,132 2,103 UBS Securities Australia Ltd Buy 2,132 2,103 UBS Securities Australia Ltd Buy 2,132 2,140 UBS Securities Australia Ltd Buy 2,132 2,140 UBS Securities Australia Ltd Buy 2,132 3,28 UBS Securities Australia Ltd Buy 2,132 3,28 UBS Securities Australia Ltd Buy 2,29 3,28 UBS Securities Australia Ltd Buy 1,581 1,785 UBS Securities Australia Ltd Buy 1,591 1,785 UBS Securities Australia Ltd Buy 2,526 67 UBS Securities Australia Ltd Buy 2,520 1,703 UBS Securities Australia Ltd Buy <th< td=""><td>JUN-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>29</td><td>3</td><td>Ordinary</td></th<>	JUN-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
UBS Securities Australia Ltd Buy 6,955 177 UBS Securities Australia Ltd Buy 6,955 177 UBS Securities Australia Ltd Buy 1,522 161 UBS Securities Australia Ltd Buy 3,93 3,66 UBS Securities Australia Ltd Buy 5,102 2,70 UBS Securities Australia Ltd Buy 5,132 3,58 UBS Securities Australia Ltd Buy 1,222 1,222 1,22 UBS Securities Australia Ltd Buy 1,232 1,23 1,62 UBS Securities Australia Ltd Buy 1,232 1,73 1,73 UBS Securities Australia Ltd Buy 2,925 3,085 1,73 1,73 UBS Securities Australia Ltd Buy 1,591 1,593 1,505 1,805 UBS Securities Australia Ltd Buy 2,785 1,401 1,73 1,73 UBS Securities Australia Ltd Buy 2,785 1,70 1,73 1,72 1,72 1,72 1,72 1,72 1,72 <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>53</td> <td>m</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buy	53	m	Ordinary
UBS Securities Australia Ltd Buy 6,955 777 UBS Securities Australia Ltd Buy 1,562 161 UBS Securities Australia Ltd Buy 20,370 2,103 UBS Securities Australia Ltd Buy 2,192 2,261 UBS Securities Australia Ltd Buy 2,192 2,261 UBS Securities Australia Ltd Buy 2,192 2,561 UBS Securities Australia Ltd Buy 1,235 1,743 UBS Securities Australia Ltd Buy 1,235 1,743 UBS Securities Australia Ltd Buy 1,535 1,783 UBS Securities Australia Ltd Buy 7,265 7,89 UBS Securities Australia Ltd Buy 7,235 1,783 UBS Securities Australia Ltd Buy 1,7315 1,783 UBS Securities Australia Ltd Buy 2,992 3,00 UBS Securities Australia Ltd Buy 2,190 3,00 UBS Securities Australia Ltd Buy 2,190 3,00 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buv	26	Q.	Ordinary
UBS Securities Australia Ltd Buy 1,520 1,61 UBS Securities Australia Ltd Buy 1,520 1,61 UBS Securities Australia Ltd Buy 20,370 2,100 UBS Securities Australia Ltd Buy 51,032 5,261 UBS Securities Australia Ltd Buy 51,032 5,261 UBS Securities Australia Ltd Buy 5,925 1,031 UBS Securities Australia Ltd Buy 5,925 3,085 UBS Securities Australia Ltd Buy 1,7315 1,785 UBS Securities Australia Ltd Buy 1,932 3,233 UBS Securities Australia Ltd Buy 1,7315 1,785 UBS Securities Australia Ltd Buy 1,531 1,785 UBS Securities Australia Ltd Buy 1,531 1,532 UBS Securities Australia Ltd Buy 2,786 6,67 UBS Securities Australia Ltd Buy 3,153 3,25 UBS Securities Australia Ltd Buy 3,153 3,153 UBS Securities Australia Ltd	hm-14	I IRS Sequities Australia ! td	Busy	6 OEE	2 :	in in
UBS Securities Australia Ltd Buy 20,370 2,100 UBS Securities Australia Ltd Buy 20,370 2,100 UBS Securities Australia Ltd Buy 2,132 2,501 UBS Securities Australia Ltd Buy 2,132 2,501 UBS Securities Australia Ltd Buy 1,222 1,401 UBS Securities Australia Ltd Buy 1,222 1,401 UBS Securities Australia Ltd Buy 1,222 1,401 UBS Securities Australia Ltd Buy 1,232 1,403 UBS Securities Australia Ltd Buy 1,731 1,785 UBS Securities Australia Ltd Buy 2,786 1,787 UBS Securities Australia Ltd Buy 1,783 3,786 UBS Securities Australia Ltd	1	Libo Consisting Augustia 14d	600	1000	, ;	, a
UBS Securities Australia Ltd Buy 2,100 UBS Securities Australia Ltd Buy 349 3,60 UBS Securities Australia Ltd Buy 51,032 2,561 UBS Securities Australia Ltd Buy 51,032 2,561 UBS Securities Australia Ltd Buy 1,232 1,273 UBS Securities Australia Ltd Buy 7,265 7,483 UBS Securities Australia Ltd Buy 7,265 3,085 UBS Securities Australia Ltd Buy 6,097 1,785 UBS Securities Australia Ltd Buy 6,097 1,785 UBS Securities Australia Ltd Buy 1,591 1,265 UBS Securities Australia Ltd Buy 1,591 1,505 UBS Securities Australia Ltd Buy 1,591 1,605 UBS Securities Australia Ltd Buy 2,1601 1,605 UBS Securities Australia Ltd Buy 2,1601 1,193 UBS Securities Australia Ltd Buy 2,160 1,605 UBS Securities Australia Ltd Buy <t< td=""><td>1</td><td>Obs Securities Australia Ltd</td><td>ing</td><td>795,1</td><td>161</td><td>Ordinary</td></t<>	1	Obs Securities Australia Ltd	ing	795,1	161	Ordinary
UBS Securities Australia Ltd Buy 5,133 36 UBS Securities Australia Ltd Buy 5,132 5,261 UBS Securities Australia Ltd Buy 1,1359 1,401 UBS Securities Australia Ltd Buy 1,1359 1,401 UBS Securities Australia Ltd Buy 1,239 1,239 1,731 UBS Securities Australia Ltd Buy 1,239 1,731 1,738 1,731 1,738 <td>41-FIZ</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>20,370</td> <td>2,100</td> <td>Ordinary</td>	41-FIZ	UBS Securities Australia Ltd	Buy	20,370	2,100	Ordinary
UBS Securities Australia Ltd Buy 5,003 5,001 UBS Securities Australia Ltd Buy 2,192 2,26 UBS Securities Australia Ltd Buy 1,590 1,401 UBS Securities Australia Ltd Buy 1,523 1,733 UBS Securities Australia Ltd Buy 1,523 1,743 UBS Securities Australia Ltd Buy 1,523 1,743 UBS Securities Australia Ltd Buy 6,507 1,743 UBS Securities Australia Ltd Buy 6,507 1,743 UBS Securities Australia Ltd Buy 1,650 1,662 UBS Securities Australia Ltd Buy 1,673 1,75 UBS Securities Australia Ltd Buy 1,650 1,60 UBS Securities Australia Ltd Buy 1,650 1,60 UBS Securities Australia Ltd Buy 2,76 6 UBS Securities Australia Ltd Buy 2,76 6 UBS Securities Australia Ltd Buy 2,76 1,60 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	349	36	Ordinary
UBS Securities Australia Ltd Buy 2,192 226 UBS Securities Australia Ltd Buy 1,390 1,401 UBS Securities Australia Ltd Buy 6,198 6,597 UBS Securities Australia Ltd Buy 1,232 1,733 UBS Securities Australia Ltd Buy 1,6907 1,743 UBS Securities Australia Ltd Buy 1,6907 1,743 UBS Securities Australia Ltd Buy 1,6907 1,743 UBS Securities Australia Ltd Buy 1,501 1,662 UBS Securities Australia Ltd Buy 1,591 1,731 UBS Securities Australia Ltd Buy 1,591 1,00 UBS Securities Australia Ltd Buy 9,1801 1,00 UBS Securities Australia Ltd Buy 6,276 6,46 UBS Securities Australia Ltd Buy 5,578 1,60 UBS Securities Australia Ltd Buy 2,469 2,546 UBS Securities Australia Ltd Buy 2,469 2,546 UBS Securities Australia Ltd <td< td=""><td>Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>51,032</td><td>5,261</td><td>Ordinary</td></td<>	Jun-14	UBS Securities Australia Ltd	Buy	51,032	5,261	Ordinary
UBS Securities Australia Ltd Buy 13,190 1,401 UBS Securities Australia Ltd Buy 6,198 6,597 UBS Securities Australia Ltd Buy 16,507 1,743 UBS Securities Australia Ltd Buy 7,265 7,89 UBS Securities Australia Ltd Buy 7,265 7,89 UBS Securities Australia Ltd Buy 17,315 1,735 UBS Securities Australia Ltd Buy 14,591 1,505 UBS Securities Australia Ltd Buy 14,591 1,505 UBS Securities Australia Ltd Buy 14,591 1,505 UBS Securities Australia Ltd Buy 1,133 1,23 UBS Securities Australia Ltd Buy 1,602 2,546 UBS Securities Australia Ltd Buy 2,4,696 2,546 UBS Securities Australia Ltd <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>2 197</td> <td>226</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buv	2 197	226	Ordinary
UBS Securities Australia Ltd Buy 6,198 6,198 UBS Securities Australia Ltd Buy 6,198 6,198 UBS Securities Australia Ltd Buy 7,265 7,49 UBS Securities Australia Ltd Buy 7,265 7,49 UBS Securities Australia Ltd Buy 7,265 7,49 UBS Securities Australia Ltd Buy 1,7315 1,785 UBS Securities Australia Ltd Buy 1,7315 1,785 UBS Securities Australia Ltd Buy 1,7315 1,785 UBS Securities Australia Ltd Buy 1,733 1,733 UBS Securities Australia Ltd Buy 1,733 1,733 UBS Securities Australia Ltd Buy 6,276 6,47 UBS Securities Australia Ltd Buy 2,696 6,47 UBS Securities Australia Ltd Buy 2,696 1,696 UBS Securities Australia Ltd Buy 2,696 1,793 UBS Securities Australia Ltd Buy 2,696 1,793 UBS Securities Australia Ltd <td< td=""><td>11n-14</td><td>ILIRS Securities Australia 11d</td><td>Britis</td><td>10 100</td><td>1 401</td><td></td></td<>	11n-14	ILIRS Securities Australia 11d	Britis	10 100	1 401	
UBS Securities Australia Ltd Buy 6.198 UBS Securities Australia Ltd Buy 16.907 1.743 UBS Securities Australia Ltd Buy 29,925 3.085 UBS Securities Australia Ltd Buy 7.265 7.86 UBS Securities Australia Ltd Buy 660 68 UBS Securities Australia Ltd Buy 16,121 1,662 UBS Securities Australia Ltd Buy 16,121 1,662 UBS Securities Australia Ltd Buy 1,133 1,23 UBS Securities Australia Ltd Buy 1,133 1,23 UBS Securities Australia Ltd Buy 1,133 1,23 UBS Securities Australia Ltd Buy 1,133 3,546 UBS Securities Australia Ltd Buy 2,4,696 2,546 UBS Securities Australia Ltd Buy 2,4,696 2,546 UBS Securities Australia Ltd Buy 2,4,696 2,546 UBS Securities Australia Ltd Buy 2,646 4,77 UBS Securities Australia Ltd Buy <td< td=""><td>1</td><td>Let I character A societies and a second</td><td>600</td><td>060,61</td><td>9</td><td>Ciditaly</td></td<>	1	Let I character A societies and a second	600	060,61	9	Ciditaly
UBS Securities Australia Ltd Buy 1,232 1,23 UBS Securities Australia Ltd Buy 1,292 1,433 UBS Securities Australia Ltd Buy 29,925 3,085 UBS Securities Australia Ltd Buy 72,655 749 UBS Securities Australia Ltd Buy 17,315 1,785 UBS Securities Australia Ltd Buy 1,591 1,785 UBS Securities Australia Ltd Buy 1,193 1,250 UBS Securities Australia Ltd Buy 1,193 1,250 UBS Securities Australia Ltd Buy 3,153 3,25 UBS Securities Australia Ltd Buy 2,466 4,75 UBS Securities Australia Ltd Buy 3,578 1,600 UBS Securities Australia Ltd Buy 4,511 4,65 UBS Securities Australia Ltd Buy	1	Opp Securities Australia Liu	had	0,198	659	Ordinary
UBS Securities Australia Ltd Buy 10,403 UBS Securities Australia Ltd Buy 7,265 3,085 UBS Securities Australia Ltd Buy 7,265 7,495 UBS Securities Australia Ltd Buy 1,7,315 1,785 UBS Securities Australia Ltd Buy 11,315 1,785 UBS Securities Australia Ltd Buy 1,193 1,234 UBS Securities Australia Ltd Buy 9,1801 9,48 UBS Securities Australia Ltd Buy 9,1801 1,234 UBS Securities Australia Ltd Buy 9,1801 9,48 UBS Securities Australia Ltd Buy 9,1801 9,77 UBS Securities Australia Ltd Buy 2,546 2,546 UBS Securities Australia Ltd Buy 2,696 2,546 UBS Securities Australia Ltd Buy 2,696 2,546 UBS Securities Australia Ltd Buy 2,646 4,791 UBS Securities Australia Ltd Buy 4,646 4,791 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	1,232	127	Ordinary
UBS Securities Australia Ltd Buy 29.925 3 085 UBS Securities Australia Ltd Buy 7,265 68 UBS Securities Australia Ltd Buy 660 68 UBS Securities Australia Ltd Buy 17,315 1,785 UBS Securities Australia Ltd Buy 16,121 1,662 UBS Securities Australia Ltd Buy 1,1931 1,505 UBS Securities Australia Ltd Buy 1,1931 1,505 UBS Securities Australia Ltd Buy 1,1931 1,505 UBS Securities Australia Ltd Buy 3,153 325 UBS Securities Australia Ltd Buy 24,696 1,607 UBS Securities Australia Ltd Buy 24,698 1,00 UBS Securities Australia Ltd Buy 24,698 1,00 UBS Securities Australia Ltd Buy 24,698 1,00 UBS Securities Australia Ltd Buy 24,698 2,546 UBS Securities Australia Ltd Buy 4,517 4,75 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buy	16,907	1,743	Ordinary
UBS Securities Australia Ltd Buy 7,265 749 UBS Securities Australia Ltd Buy 22,330 3,333 UBS Securities Australia Ltd Buy 17,315 1,785 UBS Securities Australia Ltd Buy 1,731 1,785 UBS Securities Australia Ltd Buy 1,4591 1,505 UBS Securities Australia Ltd Buy 9,1801 9,484 UBS Securities Australia Ltd Buy 9,1801 9,484 UBS Securities Australia Ltd Buy 9,1801 9,484 UBS Securities Australia Ltd Buy 6,276 6,47 UBS Securities Australia Ltd Buy 2,469 2,546 UBS Securities Australia Ltd Buy 2,469 2,546 UBS Securities Australia Ltd Buy 2,472 4,675 UBS Securities Australia Ltd Buy 2,472 4,675 UBS Securities Australia Ltd Buy 2,472 4,675 UBS Securities Australia Ltd Buy 4,646 4,772 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buy	29,925	3,085	Ordinary
UBS Securities Australia Ltd Buy 660 68 UBS Securities Australia Ltd Buy 3,330 3,333 UBS Securities Australia Ltd Buy 11,315 1,765 UBS Securities Australia Ltd Buy 14,591 1,505 UBS Securities Australia Ltd Buy 91,801 9,481 UBS Securities Australia Ltd Buy 91,801 9,481 UBS Securities Australia Ltd Buy 91,801 9,481 UBS Securities Australia Ltd Buy 9,801 9,801 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 2,646 4,99 UBS Securities Australia Ltd Buy 4,646 4,99 UBS Securities Australia Ltd Buy 4,646 4,99 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buv	7,265	749	Ordinary
UBS Securities Australia Ltd Buy 32,330 3,333 UBS Securities Australia Ltd Buy 17,315 1,785 UBS Securities Australia Ltd Buy 17,315 1,785 UBS Securities Australia Ltd Buy 91,801 9,464 UBS Securities Australia Ltd Buy 91,801 9,464 UBS Securities Australia Ltd Buy 91,801 9,464 UBS Securities Australia Ltd Buy 9,89 102 UBS Securities Australia Ltd Buy 3,153 355 UBS Securities Australia Ltd Buy 3,69 5,66 UBS Securities Australia Ltd Buy 3,47 3,71 UBS Securities Australia Ltd Buy 9,477 3,71 UBS Securities Australia Ltd Buy 10,602 1,033 UBS Securities Australia Ltd Buy 4,472 4,475 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Bu	110-14	I BS Securities Australia I to	Bira	999	0.9	Ordingo
UBS Securities Australia Ltd Buy 17,313 UBS Securities Australia Ltd Buy 15,121 1,662 UBS Securities Australia Ltd Buy 16,121 1,662 UBS Securities Australia Ltd Buy 1,193 1,133 1,234 UBS Securities Australia Ltd Buy 3,153 3,55 3,55 UBS Securities Australia Ltd Buy 8,00 3,153 3,55 1,00 UBS Securities Australia Ltd Buy 8,00 3,153 3,55 6,00 UBS Securities Australia Ltd Buy 24,696 2,546 9,7 1,00 UBS Securities Australia Ltd Buy 24,696 2,546 4,79 UBS Securities Australia Ltd Buy 2,4,696 2,546 4,79 UBS Securities Australia Ltd Buy 2,646 4,79 1,606 UBS Securities Australia Ltd Buy 4,547 4,79 1,606 UBS Securities Australia Ltd Buy 4,547 4,79 1,23 UBS Securities Australia Ltd	17	Del Compies Americal 14		055.55	200	, i
UBS Securities Australia Ltd Buy 1,781 1,785 UBS Securities Australia Ltd Buy 16,121 1,602 UBS Securities Australia Ltd Buy 1,193 1,233 UBS Securities Australia Ltd Buy 1,193 1,23 UBS Securities Australia Ltd Buy 3,153 3,25 UBS Securities Australia Ltd Buy 989 102 UBS Securities Australia Ltd Buy 340 35 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 340 35 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 39,710 37 UBS Securities Australia Ltd Buy 3,573 1,606 UBS Securities Australia Ltd Buy 4,511 4,65 UBS Securities Australia Ltd Buy 4,511 4,65 UBS Securities Australia Ltd Buy 4,511 4,65 UBS Securities Australia Ltd Buy		The Committee Augustian Like	(ng	52,330	5,533	Ordinary
UBS Securities Australia Ltd Buy 16,21 1,62 UBS Securities Australia Ltd Buy 1,59 1,50 UBS Securities Australia Ltd Buy 1,133 1,23 2,24 64 1,23 1,23 2,24 64 1,23 2,24 66 1,23 2,24 66 1,23 2,24 66 1,23 2,24 66 2,24 2,24 66 2,24 2,24 66 2,24 2,24 2,24 66 2,24 2,24	- In	UBS Securities Australia Ltd	Buy	17,315	1,785	Ordinary
UBS Securities Australia Ltd Buy 14,591 1,505 UBS Securities Australia Ltd Buy 9,484 1,484 UBS Securities Australia Ltd Buy 3,153 3,23 UBS Securities Australia Ltd Buy 9,891 102 UBS Securities Australia Ltd Buy 989 102 UBS Securities Australia Ltd Buy 340 35 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 39,77 10 UBS Securities Australia Ltd Buy 2,466 479 UBS Securities Australia Ltd Buy 2,576 1,606 UBS Securities Australia Ltd Buy 4,646 479 UBS Securities Australia Ltd Buy 4,617 475 UBS Securities Australia Ltd Buy 4,727 461 UBS Securities Australia Ltd Buy 4,614 4,75 UBS Securities Australia Ltd Buy <	Jun-14	UBS Securities Australia Ltd	Buy	16,121	1,662	Ordinary
UBS Securities Australia Ltd Buy 91,801 9,464 UBS Securities Australia Ltd Buy 1,193 123 UBS Securities Australia Ltd Buy 6,276 647 UBS Securities Australia Ltd Buy 989 102 UBS Securities Australia Ltd Buy 2,666 6 UBS Securities Australia Ltd Buy 2,666 2,546 UBS Securities Australia Ltd Buy 2,469 2,546 UBS Securities Australia Ltd Buy 2,477 977 UBS Securities Australia Ltd Buy 2,646 4,79 UBS Securities Australia Ltd Buy 2,646 4,79 UBS Securities Australia Ltd Buy 2,646 4,79 UBS Securities Australia Ltd Buy 4,646 4,79 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	14.591	1.505	Ordinary
UBS Securities Australia Ltd Buy 1,750 1,723 1,724 UBS Securities Australia Ltd Buy 3,153 3,154 3,154 3,154 3,100 3,154 3,154 3,154 3,154 3,154 3,154 3,154 3,154 3,154 3,154 3,154 </td <td>110-14</td> <td>UBS Securities Australia Ltd</td> <td>B</td> <td>01 901</td> <td>0 454</td> <td>i de dipol</td>	110-14	UBS Securities Australia Ltd	B	01 901	0 454	i de dipol
UBS Securities Australia Ltd Buy 1,193 123 UBS Securities Australia Ltd Buy 6,276 647 UBS Securities Australia Ltd Buy 340 355 UBS Securities Australia Ltd Buy 340 355 UBS Securities Australia Ltd Buy 347 37 UBS Securities Australia Ltd Buy 347 37 UBS Securities Australia Ltd Buy 3477 37 UBS Securities Australia Ltd Buy 2,646 4.79 UBS Securities Australia Ltd Buy 4,646 4.79 UBS Securities Australia Ltd Buy 2,646 4.79 UBS Securities Australia Ltd Buy 4,646 4.79 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 4,472 4,615 UBS Securities Australia Ltd Buy 4,472 4,615 UBS Securities Australia Ltd Buy 4,472	1	I IDC Commission Assertation 1-th	(50	100,16	40.	ordinal y
UBS Securities Australia Ltd Buy 3.153 3.25 UBS Securities Australia Ltd Buy 989 10.2 UBS Securities Australia Ltd Buy 989 10.2 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 24,77 100 UBS Securities Australia Ltd Buy 2,646 4.79 UBS Securities Australia Ltd Buy 2,646 4.79 UBS Securities Australia Ltd Buy 4,647 4.79 UBS Securities Australia Ltd Buy 4,647 4.79 UBS Securities Australia Ltd Buy 4,77 4.61 UBS Securities Australia Ltd Buy 4,77 4.71 UBS Securities Australia Ltd Buy 4,77 4.61 UBS Securities Australia Ltd Buy 4,77 4.71 UBS Securities Australia Ltd Buy	ti-It	UBS Securities Australia Ltd	Buy	1,193	123	Ordinary
UBS Securities Australia Ltd Buy 6,276 647 UBS Securities Australia Ltd Buy 389 102 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 97 10 UBS Securities Australia Ltd Buy 97 10 UBS Securities Australia Ltd Buy 9,7 10 UBS Securities Australia Ltd Buy 2,477 977 UBS Securities Australia Ltd Buy 2,646 4.79 UBS Securities Australia Ltd Buy 4,646 4.79 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 1,763 1,723 UBS Securities Australia Ltd Buy 1,476 1,522 UBS Securities Australia Ltd Buy 1,245	Jun-14	UBS Securities Australia Ltd	Buy	3,153	325	Ordinary
UBS Securities Australia Ltd Buy 989 102 UBS Securities Australia Ltd Buy 54 56 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 24,70 10 UBS Securities Australia Ltd Buy 2,578 1,606 UBS Securities Australia Ltd Buy 2,578 1,606 UBS Securities Australia Ltd Buy 4,646 4,79 UBS Securities Australia Ltd Buy 4,646 4,79 UBS Securities Australia Ltd Buy 4,611 4,65 UBS Securities Australia Ltd Buy 4,71 4,61 UBS Securities Australia Ltd Buy 4,72 4,61 UBS Securities Australia Ltd Buy 4,77 4,61 UBS Securities Australia Ltd Buy 4,77 4,61 UBS Securities Australia Ltd Buy 4,77 4,75 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	6.276	647	Ordinary
UBS Securities Australia Ltd Buy 340 35 UBS Securities Australia Ltd Buy 2,696 2,546 UBS Securities Australia Ltd Buy 10,602 1,093 UBS Securities Australia Ltd Buy 9,7 10 UBS Securities Australia Ltd Buy 2,04 1,093 UBS Securities Australia Ltd Buy 2,04 1,7 UBS Securities Australia Ltd Buy 4,646 4,7 UBS Securities Australia Ltd Buy 4,511 465 UBS Securities Australia Ltd Buy 4,511 465 UBS Securities Australia Ltd Buy 4,772 461 UBS Securities Australia Ltd Buy 1,263 1,783 UBS Securities Australia Ltd Buy 1,2,45 1,522 UBS Securities Australia Ltd Buy 5,	Jun-14	UBS Securities Australia Ltd	Buv	080	102	Ordinan
UBS Securities Australia Ltd Buy 54 55 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 397 10 UBS Securities Australia Ltd Buy 9470 977 UBS Securities Australia Ltd Buy 2,646 4,79 UBS Securities Australia Ltd Buy 4,646 4,79 UBS Securities Australia Ltd Buy 4,646 4,79 UBS Securities Australia Ltd Buy 4,646 4,79 UBS Securities Australia Ltd Buy 4,647 4,61 UBS Securities Australia Ltd Buy 4,617 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 1,769 1,769 UBS Securities Australia Ltd Buy 5,29 5,79 UBS Securities Australia Ltd Buy	14	LIBO Contribute Australia 114	Bury	200	20 1	Cicinary
UBS Securities Australia Ltd Buy 24,696 2,64 UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 10,662 1,033 UBS Securities Australia Ltd Buy 9,477 9,77 UBS Securities Australia Ltd Buy 1,578 1,606 UBS Securities Australia Ltd Buy 4,511 465 UBS Securities Australia Ltd Buy 1,769 1,789 UBS Securities Australia Ltd Buy 1,745 1,283 UBS Securities Australia Ltd Buy 1,245 1,283 UBS Securities Australia Ltd Buy 1,248 1,283 UBS Securities Australia Ltd Buy		ODS SECULIONS HAND FOR	, and	340	cr.	Ordinary
UBS Securities Australia Ltd Buy 24,696 2,546 UBS Securities Australia Ltd Buy 2,690 1,033 UBS Securities Australia Ltd Buy 16,602 1,033 UBS Securities Australia Ltd Buy 2,047 977 UBS Securities Australia Ltd Buy 2,047 27 UBS Securities Australia Ltd Buy 4,646 4.79 UBS Securities Australia Ltd Buy 4,647 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 4,472 4,61 UBS Securities Australia Ltd Buy 1,769 1,769 UBS Securities Australia Ltd Buy 1,2,453 1,522 UBS Securities Australia Ltd Buy 1,2,453 1,522 UBS Securities Australia Ltd Buy 7,498 7,799 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy	-14 -14	UBS Securities Australia Ltd	Buy	58	9	Ordinary
UBS Securities Australia Ltd Buy 10 UBS Securities Australia Ltd Buy 10,602 UBS Securities Australia Ltd Buy 15,778 1,606 UBS Securities Australia Ltd Buy 15,778 1,606 UBS Securities Australia Ltd Buy 204 21 UBS Securities Australia Ltd Buy 4,646 479 UBS Securities Australia Ltd Buy 4,771 465 UBS Securities Australia Ltd Buy 4,772 461 UBS Securities Australia Ltd Buy 4,772 461 UBS Securities Australia Ltd Buy 1,769 2,038 UBS Securities Australia Ltd Buy 1,769 1,738 UBS Securities Australia Ltd Buy 1,749 1,283 UBS Securities Australia Ltd Buy 5,29 5,70 UBS Securities Australia Ltd Buy 1,2,445 1,283 UBS Securities Australia Ltd Buy 7,498 7,799 UBS Securities Australia Ltd Buy 7,498 7,790	Jun-14	UBS Securities Australia Ltd	Buy	24,696	2,546	Ordinary
UBS Securities Australia Ltd Buy 10,602 1,093 UBS Securities Australia Ltd Buy 9,477 977 UBS Securities Australia Ltd Buy 204 27 UBS Securities Australia Ltd Buy 4,646 479 UBS Securities Australia Ltd Buy 4,641 479 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 1,578 1,522 UBS Securities Australia Ltd Buy 1,4763 1,522 UBS Securities Australia Ltd Buy 1,2463 1,522 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,560 7,79 UBS Securities Australia Ltd Buy <	Jun-14	UBS Securities Australia Ltd	Buy	26	10	Ordinary
UBS Securities Australia Ltd Buy 9.477 9.77 UBS Securities Australia Ltd Buy 15.78 1.606 UBS Securities Australia Ltd Buy 2.64 4.79 UBS Securities Australia Ltd Buy 4,646 4.79 UBS Securities Australia Ltd Buy 4,611 465 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 4,475 461 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,	Jun-14	UBS Securities Australia Ltd	Buy	10.602	1 093	Ordinary
UBS Securities Australia Ltd Buy 15,578 1,606 UBS Securities Australia Ltd Buy 204 27 UBS Securities Australia Ltd Buy 4,646 479 UBS Securities Australia Ltd Buy 4,611 465 UBS Securities Australia Ltd Buy 4,711 465 UBS Securities Australia Ltd Buy 4,772 461 UBS Securities Australia Ltd Buy 1,522 177 UBS Securities Australia Ltd Buy 1,4763 1,522 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 5,229 570 UBS Securities Australia Ltd Buy 5,229 570 UBS Securities Australia Ltd Buy 5,245 570 UBS Securities Australia Ltd Buy 5,229 570 UBS Securities Australia Ltd Buy 7,498 7,790 UBS Securities Australia Ltd Buy 7,600 7,800 UBS Securities Australia Ltd Buy <td< td=""><td>1 m-1 A</td><td>LIBC Counties Australia Ltd</td><td>6.0</td><td>200/01</td><td></td><td>1</td></td<>	1 m-1 A	LIBC Counties Australia Ltd	6.0	200/01		1
UBS Securities Australia Ltd Buy 1,50/B 1,50/B UBS Securities Australia Ltd Buy 7,57/B 4,79 UBS Securities Australia Ltd Buy 4,646 4,79 UBS Securities Australia Ltd Buy 4,511 465 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 1,769 1,769 UBS Securities Australia Ltd Buy 1,4763 1,522 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,496 7,79 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,698 773 UBS Securities Australia Ltd Buy 7,698 773 UBS Securities Australia Ltd Buy		D 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	600	(11)	,,,,	Oldina.y
UBS Securities Australia Ltd Buy 204 21 UBS Securities Australia Ltd Buy 4,646 479 UBS Securities Australia Ltd Buy 4,511 465 UBS Securities Australia Ltd Buy 4,511 465 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 1,676 2,038 UBS Securities Australia Ltd Buy 1,476 1,522 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 1,245 1,283 UBS Securities Australia Ltd Buy 5,229 5,70 UBS Securities Australia Ltd Buy 7,498 7,73 UBS Securities Australia Ltd Buy 7,498 7,73 UBS Securities Australia Ltd Buy 7,600 7,780 UBS Securities Australia Ltd Buy 7,600 7,780 UBS Securities Australia Ltd Buy 7,600 7,800 UBS Securities Australia Ltd Buy	1	ODS SECULINES AUSINAINA LIG	any	8/5,51	909,1	Ordinary
UBS Securities Australia Ltd Buy 4,646 479 UBS Securities Australia Ltd Buy 281 29 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 4,472 461 UBS Securities Australia Ltd Buy 1,522 1,7 UBS Securities Australia Ltd Buy 1,2,63 1,522 UBS Securities Australia Ltd Buy 1,2,45 1,522 UBS Securities Australia Ltd Buy 1,2,45 1,283 UBS Securities Australia Ltd Buy 5,529 5,70 UBS Securities Australia Ltd Buy 7,498 7,79 UBS Securities Australia Ltd Buy 7,498 7,79 UBS Securities Australia Ltd Buy 7,698 773	Jun-14	UBS Securities Australia Ltd	Buy	204	21	Ordinary
UBS Securities Australia Ltd Buy 281 29 UBS Securities Australia Ltd Buy 4,511 465 UBS Securities Australia Ltd Buy 4,727 461 UBS Securities Australia Ltd Buy 1,769 1,78 UBS Securities Australia Ltd Buy 1,3,69 1,232 UBS Securities Australia Ltd Buy 1,2,45 1,283 UBS Securities Australia Ltd Buy 1,2,45 1,283 UBS Securities Australia Ltd Buy 5,29 5,70 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 775 UBS Securities Australia Ltd Buy 8,60 860	Jun-14	UBS Securities Australia Ltd	Buy	4.646	479	Ordinary
UBS Securities Australia Ltd Buy 4,511 465 UBS Securities Australia Ltd Buy 4,712 461 UBS Securities Australia Ltd Buy 165 17 UBS Securities Australia Ltd Buy 19,769 2,038 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 1,245 1,522 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 7,760 800	lun-14	UBS Securities Australia (1d	Sic	281	٥	Ordinary
UBS Securities Australia Ltd	2	LIBC Counting Australia 1+4	600	7 2 2 2	27	A DE LOS
UBS Securities Australia Ltd Buy 4.72 461 UBS Securities Australia Ltd Buy 1.67 1.7 UBS Securities Australia Ltd Buy 1.9,769 2,038 UBS Securities Australia Ltd Buy 1,4,763 1,522 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 5,529 5,70 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 7,760 800		ODS Securities Australia Cru	hna	4,01	402	Ordinary
UBS Securities Australia Ltd Buy 165 17 UBS Securities Australia Ltd Buy 19,769 2,038 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 6,124 631 UBS Securities Australia Ltd Buy 12,445 1,283 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,498 7750 UBS Securities Australia Ltd Buy 7,698 773 UBS Securities Australia Ltd Buy 7,698 773	Jun-14	UBS securities Australia Ltd	Buy	4,472	461	Ordinary
UBS securities Australia Ltd Buy 19,769 2,038 USS Securities Australia Ltd Buy 1,522 1,522 USS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 1,2,45 1,283 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 7,760 800	Jun-14	UBS Securities Australia Ltd	Buy	165	17	Ordinary
UBS Securities Australia Ltd Buy 14,763 1,522 UBS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 12,445 1,283 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 1,786 773	Jun-14	UBS Securities Australia Ltd	Buv	19.769	2.038	Ordinary
USS Securities Australia Ltd Buy 6,121 631 USS Securities Australia Ltd Buy 12,445 1,283 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 5,529 773 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 18,033 18,65	4 m-14	I IRS Securities Australia Ltd	A G	14.762	1 522	Ordinan
USS Securities Australia Ltd Buy 6,121 631 UBS Securities Australia Ltd Buy 11,245 1,283 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 1,760 800	1	COS Securities Australia Liu	, and	14,703	775,1	Ordinary
UBS Securities Australia Ltd Buy 12,445 1,283 UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,760 800 UBS Securities Australia Ltd Buy 1,803 1,865	Jun-14	UBS Securities Australia Ltd	Buy	6,121	631	Ordinary
UBS Securities Australia Ltd Buy 5,529 570 UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,750 800 UBS Securities Australia Ltd Buy 18,073 18,65	Jun-14	UBS Securities Australia Ltd	Buy	12,445	1,283	Ordinary
UBS Securities Australia Ltd Buy 7,498 773 UBS Securities Australia Ltd Buy 7,750 800 UBS Securities Australia Ltd Buy 18,003 1860	Jun-14	UBS Securities Australia Ltd	Buy	5,529	570	Ordinary
UBS Securities Australia Ltd Buy 7,750 800 UBS Securities Australia Ltd Buy 18,003 1,866	Jun-14	UBS Securities Australia Ltd	Buv	7.498	773	Ordinary
UBS Securities Australia Ltd Buy 18-56	11-14	ILIRS Securities Australia ! td	Bliv	7 760	000	,
IOBS Securities Australia Ltd IBuv 1856		מבו הבינה היינה היינה בינה			×	Contract of the Contract of th
000'1		The Charles of the Control		20.44	000	Ordinary

	UBS Securities Australia 11d	VI B	58,016	5,981	Ordinary
1-JUN-14	מוש היושה ויישה ויישה היושה היושה	6			
1-Jun-14	UBS Securities Australia Ltd	Buy	3,936	406	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,860	501	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,715	383	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buv	4.171	430	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buv	14.812	1.527	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Btiv	101	F	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Biv	2 801	308	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	11 291	1,164	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Biv	27E 6	345	Ordinary
1-Jun-14	UBS Securities Australia I td	Bis	278	a a	Ordinary
14-011	I IRS Securities Australia and	Barr	97.5	755	Ordinary
1-lin-14	LIBS Securities Australia 11d	Buy	0370	200	Ordinary
1-lin-14	I IRS Sequities Australia I to	Brox	7 271	750	Collings
1-hin-14	LIRS Securities Australia Ltd	S S S S S S S S S S S S S S S S S S S	710 111	11 445	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Birv	3 822	394	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	29	٣	Ordinary
Jun-14	UBS Securities Australia Ltd	Biv	29) m	Ordinary
1-Jun-14	UBS Securities Australia I td	BIN	101	2 0	Ordinan
1-lun-14	UBS Securities Australia I td	Birv	2 5.41	262	Ordinan
1-Jun-14	UBS Securities Australia ! td	∑ B	2 978	307	Ordinary
1-Jun-14	UBS Securities Australia Itd	Bitv	456	47	Ordinary
-lun-14	UBS Securities Australia (1d	Biv	1 698	175	Ordinary
1-1110-14	I IRS Securities Australia I td	Brie	6.480	899	Ordinany
1-lin-14	I IBS Securities Australia I td	2 8	0,700	-	Codingra
1-hm-14	URS Securities Australia Ind	Big	2 00	- 6	Ordinary
1-1 10-14	FRS Countries Australia 14d	600	0,000	700,	Ordinary Ordinary
1-lin-14	I IRS Securities Australia 11d	200	531.5	200	Ordinary
110.14	I IRS Sequenties Australia 14d	200	1040	100	Codingry
1	I IDC Cog mittor Auctralia 14d		2000	2 5	Ordinary
11 100 14	U.D.C. Constitution Australia Ltd	sna ens	700/7	7/7	Orginary
# 1 Inc	Opp Securities Australia Lid	pon	67	٦ .	Ordinary
- Jun - 14	UBS Securities Australia Ltd	Buy	2,058	2/4	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	243	52	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	2,910	300	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	3,366	347	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	8,109	836	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	2,221	525	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	87	6	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,153	325	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Birv	4171	430	Ordinary
1-lun-14	LIBS Securities Australia Ltd	A B	2347	242	Ordinary
1 1 1	I IBC Con mition Australia 144	in a	745'7	747	Ordinary
t i	OBS Securities Australia Liu	and a	01-	7	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	766	79	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	3,638	375	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	3,638	375	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	293	28	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	4,074	420	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	1.678	173	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	29	٣	Ordinary
lin-14	I IBS Securities Australia I to	Bis	300	7	Ording
1 1 1	De Consider A material		62	1	o consist
Jul - 14	OBS Securities Australia Ltd	hog	401	5	Ordinary
11-Jun-14	OBS Securities Australia Ltd	Buy	4,152	478	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	2,474	255	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	194	20	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	53	3	Ordinary
11-Jun-14	UBS Securities Australia Ltd	Buy	553	57	Ordinary
11-Jun-14	UBS Securities Australia I to	Risy	56	r	Ordinary
1-hm-14	LIBS Securities Australia 14d	200	000	7	ordina.
	I IDC Commission A control and Life	, and	67	7	Cruitaly
	UBS Securities Australia Liu	ang	67	5	Ordinary
JUD-14	UBS Securities Australia Ltd	Buy	67	3	Ordinary
- 1	UBS Securities Australia Ltd	Buy	56	3	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buy	29	r	Carrier
				1	Gunary
1-Jun-14	IUBS Securities Australia Ltd	Buv	3.114	321	Ordinary

695 Securities Australia LID BUY 21,038 685 Securities Australia LID BUY 7,044 685 Securities Australia LID BUY 7,644 2,17,17 105 Securities Australia LID BUY 7,644 1,019 1,1 105 Securities Australia LID BUY 5,848 1,019 1,1 106 Securities Australia LID BUY 5,849 2,175 107 Securities Australia LID BUY 5,849 2,471 108 Securities Australia LID BUY 5,849 2,471 108 Securities Australia LID BUY 1,520 1,520 108 Securities Australia LID BUY 1,670 1,520 108 Securities Australia LID <t< th=""><th>1-Jun-14</th><th>I IDC Contration Australia Ltd</th><th>7.00</th><th>c.</th><th></th><th></th></t<>	1-Jun-14	I IDC Contration Australia Ltd	7.00	c.		
USS Securities Australia Ltd Buy 1999 1009	un-14	USS Securities Australia Itd	Buly	21 008	2 17E	Ordinary
March March May 1964 1967 1968 1069 1069 1069 1068 1069 1068	Un-14	UBS Securities Australia Ltd	Buv	10	2,1/3	Ordinary
March March May 17,864 827 1019 March May 1,284 1019 1019 March March May 1,287 144 March March March May 1,287 142 March March March May 1,287 1,280 March March March May 1,280 1,220 March March March May 1,220 March March March May 1,220 March March March March May 1,220 March March March March May 1,220 March	Jun-14	UBS Securities Australia Ltd	Buv	10		Ordinary
Libit Securities Australia Ltd Buy 5,884 1,093 LibS Securities Australia Ltd Buy 5,884 1,093 LibS Securities Australia Ltd Buy 1,399 1,441 LibS Securities Australia Ltd Buy 6,014 6,014 LibS Securities Australia Ltd Buy 6,014 6,014 LibS Securities Australia Ltd Buy 6,014 1,516 LibS Securities Australia Ltd Buy 1,616 1,516 LibS Securities Australia Ltd Buy 1,616 1,516 LibS Securities Australia Ltd Buy 1,620 1,620 LibS Securities Australia L	Un-14	UBS Securities Australia 11d	Bir	7 964	100	Ordinary
LOSS Securities Australia Ltd Buy 5,481 5,581 144 LOSS Securities Australia Ltd Buy 6,79 1,239 1,44 LOSS Securities Australia Ltd Buy 6,70 6,50 1 1 LOSS Securities Australia Ltd Buy 6,70 1,50 1 1 1 LOSS Securities Australia Ltd Buy 1,80 1,00 1	lun-14	UBS Securities Australia Ltd	Buv	9 884	0101	Ordinary
UBS Securities Australia Ltd Bly	Jun-14	UBS Securities Australia Ltd	Buy	5.481	565	Ordinary
UBS Securities Australia Ltd Bly	Jun-14	UBS Securities Australia Ltd	Buv	1 397	144	Ordinary
UBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia Ltd	Buy	2.396	247	Ordinary
UBS Securities Australia Ltd Bby 10 14 LBS Securities Australia Ltd Bby 190 14 LBS Securities Australia Ltd Bby 190 14 LBS Securities Australia Ltd Bby 180 190 14 LBS Securities Australia Ltd Bby 180 180 180 180 LBS Securities Australia Ltd Bby 180 <t< td=""><td>Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>6,014</td><td>620</td><td>Ordinary</td></t<>	Jun-14	UBS Securities Australia Ltd	Buy	6,014	620	Ordinary
UBS Securities Australia Ltd Buy 9.89 1.02 UBS Securities Australia Ltd Buy 9.81 1.02 UBS Securities Australia Ltd Buy 1.93 1.38 1.42.00 UBS Securities Australia Ltd Buy 1.93 1.39 1.51 1.00 UBS Securities Australia Ltd Buy 1.62 7.01 1.10 1.00 UBS Securities Australia Ltd Buy 1.62 7.01 1.57 1.00 UBS Securities Australia Ltd Buy 5.01 2.03 1.62 7.01 UBS Securities Australia Ltd Buy 5.02 2.02 1.62 7.00 UBS Securities Australia Ltd Buy 5.02 2.02 1.62 7.00 UBS Securities Australia Ltd Buy 6.02 2.02 6.24 1.00 UBS Securities Australia Ltd Buy 6.02 2.02 6.24 1.00 UBS Securities Australia Ltd Buy 6.02 2.02 6.24 0.00 UBS Securities Australia Ltd Buy 6.02 2.02 6.24 0.00 UBS Securities Australia Ltd Buy 6.02 0.02 1.00 0.00 UBS Securities Australia Ltd Buy 6.02 0.00	Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
UBS Securities Australia Ltd Buy 19.981 1,0.20 UBS Securities Australia Ltd Buy 1,9.10 1,5.10 <t< td=""><td>Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>39</td><td>4</td><td>Ordinary</td></t<>	Jun-14	UBS Securities Australia Ltd	Buy	39	4	Ordinary
UBS Securities Australia Ltd Buy 118,128 14,20 UBS Securities Australia Ltd Buy 1,516 1,07 1,516 UBS Securities Australia Ltd Buy 1,07 1,516 1,00 UBS Securities Australia Ltd Buy 1,620 1,572 1,572 UBS Securities Australia Ltd Buy 1,620 1,572 1,000 UBS Securities Australia Ltd Buy 1,620 1,672 1,000 UBS Securities Australia Ltd Buy 6,221 1,000 1,641 UBS Securities Australia Ltd Buy 6,222 1,642 1,654 UBS Securities Australia Ltd Buy 6,223 1,642 1,641 UBS Securities Australia Ltd Buy 1,642 1,642 1,641 UBS Securities Australia Ltd Buy 1,642 1,642 1,643 UBS Securities Australia Ltd Buy 1,640 1,640 1,640 UBS Securities Australia Ltd Buy 1,640 1,640 1,640 UBS Securities Australia Ltd <td>In-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>186'6</td> <td>1,029</td> <td>Ordinary</td>	In-14	UBS Securities Australia Ltd	Buy	186'6	1,029	Ordinary
UBS Securities Australia Ltd Bluy 14,705 1,516 UBS Securities Australia Ltd Bluy 1,507 1,516 UBS Securities Australia Ltd Bluy 1,620 2.23 UBS Securities Australia Ltd Bluy 1,620 1,520 UBS Securities Australia Ltd Bluy 1,620 1,520 UBS Securities Australia Ltd Bluy 1,620 1,627 UBS Securities Australia Ltd Bluy 1,620 1,627 UBS Securities Australia Ltd Bluy 6,628 6,621 UBS Securities Australia Ltd Bluy 6,628 6,621 UBS Securities Australia Ltd Bluy 6,628 6,621 UBS Securities Australia Ltd Bluy 2,841 1,691 UBS Securities Australia Ltd Bluy 1,862 1,691 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buy	138,128	14,240	Ordinary
UBS Securities Australia Ltd Bluy 107 11 UBS Securities Australia Ltd Bluy 223 5.23 UBS Securities Australia Ltd Bluy 1.672 1.672 UBS Securities Australia Ltd Bluy 1.6529 1.672 UBS Securities Australia Ltd Bluy 1.6529 1.672 UBS Securities Australia Ltd Bluy 6,262 1.672 UBS Securities Australia Ltd Bluy 6,262 1.672 UBS Securities Australia Ltd Bluy 6,263 1.672 UBS Securities Australia Ltd Bluy 6,263 1.672 UBS Securities Australia Ltd Bluy 6,263 1.672 UBS Securities Australia Ltd Bluy 1.673 1.648 UBS Securities Australia Ltd Bluy 1.648 1.620 UBS Securities Australia Ltd Bluy 1.648 1.648 UBS Securities Australia Ltd Bluy 1.648 1.648 UBS Securities Australia Ltd Bluy 1.657 1.111 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buy	14,705	1,516	Ordinary
UBS Securities Australia Ltd Blay 5,23 5,23 UBS Securities Australia Ltd Blay 1,872 5,23 UBS Securities Australia Ltd Blay 1,872 1,579 UBS Securities Australia Ltd Blay 1,872 1,570 UBS Securities Australia Ltd Blay 9,589 8,340 UBS Securities Australia Ltd Blay 9,512 1,000 UBS Securities Australia Ltd Blay 6,282 6,481 UBS Securities Australia Ltd Blay 2,583 1,548 UBS Securities Australia Ltd Blay 2,28,304 1,567 UBS Securities Australia Ltd Blay 2,883 1,567 UBS Securities Australia Ltd Blay 2,833 1,567 UBS Securities Australia Ltd Blay 1,895 1,480 UBS Securities Australia Ltd Blay 1,897 1,480 UBS Securities Australia Ltd Blay 1,897 1,480 UBS Securities Australia Ltd Blay 1,897 1,114 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buy	107	-1	Ordinary
UBS Securities Australia Ltd Bluy 5.073 5.02 UBS Securities Australia Ltd Bluy 18.259 1.672 UBS Securities Australia Ltd Bluy 18.259 1.672 UBS Securities Australia Ltd Bluy 9.514 1.0200 UBS Securities Australia Ltd Bluy 9.514 1.0200 UBS Securities Australia Ltd Bluy 9.514 1.0200 UBS Securities Australia Ltd Bluy 6.233 6.241 UBS Securities Australia Ltd Bluy 6.241 1.051 UBS Securities Australia Ltd Bluy 1.7106 1.750 UBS Securities Australia Ltd Bluy 1.750 1.750 UBS Securities Australia Ltd Bluy 1.750 1.750 UBS Securities Australi	Jun-14	UBS Securities Australia Ltd	Buy	243	25	Ordinary
UBS Securities Australia Ltd Bluy 18,624 1,520 UBS Securities Australia Ltd Buy 16,256 1,6270 UBS Securities Australia Ltd Buy 10,257 1,6270 UBS Securities Australia Ltd Buy 6,262 6,262 UBS Securities Australia Ltd Buy 97,514 10,000 UBS Securities Australia Ltd Buy 238,487 24,582 UBS Securities Australia Ltd Buy 278,304 24,582 UBS Securities Australia Ltd Buy 238,487 (15,640) UBS Securities Australia Ltd Buy 238,487 (15,250) UBS Securities Australia Ltd Buy 278,304 (15,250) UBS Securities Australia Ltd Buy 28,333 (14,25) UBS Securities Australia Ltd Buy 10,300 (15,250) UBS Securities Australia Ltd Buy 10,300 (14,250) UBS Securities Australia Ltd Buy 10,300 (14,250) UBS Securities Australia Ltd Buy 10,300 (14,40)	Jun-14	UBS Securities Australia Ltd	Buy	5,073	523	Ordinary
UBS Securities Australia Ltd Bly 16.756 UBS Securities Australia Ltd Bly 16.756 UBS Securities Australia Ltd Bly 6.241 UBS Securities Australia Ltd Bly 6.243 UBS Securities Australia Ltd Bly 6.243 UBS Securities Australia Ltd Bly 6.243 UBS Securities Australia Ltd Bly 7.176 UBS Securities Australia Ltd Bly 7.176 UBS Securities Australia Ltd Bly 1.176 UBS Securities Australia Ltd Bly	Jun-14	UBS Securities Australia Ltd	Buy	18,624	1,920	Ordinary
UBS Securities Australia Ltd Sell 125,547 (1,670) UBS Securities Australia Ltd Buy 6,262 6,44 UBS Securities Australia Ltd Buy 6,262 6,241 UBS Securities Australia Ltd Buy 6,552 6,241 UBS Securities Australia Ltd Buy 60,527 6,241 UBS Securities Australia Ltd Buy 60,527 6,241 UBS Securities Australia Ltd Buy 60,527 6,241 UBS Securities Australia Ltd Buy 278,304 28,522 UBS Securities Australia Ltd Buy 278,304 28,505 UBS Securities Australia Ltd Buy 278,304 28,505 UBS Securities Australia Ltd Buy 17,108 28,207 UBS Securities Australia Ltd Buy 18,078 18,290 UBS Securities Australia Ltd Buy 108,305 11,114 UBS Securities Australia Ltd Buy 108,305 11,104 UBS Securities Australia Ltd Buy 108,305 11,104 UBS Securities	Jun-14	UBS Securities Australia Ltd	Buy	16,269	1,672	Ordinary
UBS Securities Australia Ltd Bluy 89,089 89,380 UBS Securities Australia Ltd Buy 97,517 10,000 UBS Securities Australia Ltd Buy 6,262 6,44 UBS Securities Australia Ltd Buy 6,262 6,44 UBS Securities Australia Ltd Buy 228,437 2,458 UBS Securities Australia Ltd Sell 175,1987 2,645 UBS Securities Australia Ltd Sell 17,1987 2,645 UBS Securities Australia Ltd Sell 17,108 1,617 UBS Securities Australia Ltd Buy 1,708 1,617 UBS Securities Australia Ltd Buy 1,87 1,617 UBS Securities Australia Ltd Buy 1,627 1,627 UBS Securities Australia Ltd Buy 1,657 1,627 UBS Securities Australia Lt	Jun-14	UBS Securities Australia Ltd	Sell	123,547	(12,676)	Ordinary
UBS Securities Australia Ltd Bluy 97,514 10,000 UBS Securities Australia Ltd Buy 6,721 6,241 UBS Securities Australia Ltd Buy 6,721 6,241 UBS Securities Australia Ltd Buy 6,721 6,241 UBS Securities Australia Ltd Buy 2,78,304 28,824 UBS Securities Australia Ltd Sell 15,985 (15,48) UBS Securities Australia Ltd Buy 2,78,304 28,824 UBS Securities Australia Ltd Buy 2,78,304 2,88,24 UBS Securities Australia Ltd Buy 16,486 1,697 UBS Securities Australia Ltd Buy 16,486 1,697 UBS Securities Australia Ltd Buy 1,7108 1,142 UBS Securities Australia Ltd Buy 1,897 1,997 UBS Securities Australia Ltd Buy 1,897 1,997 UBS Securities Australia Ltd Buy 1,867 1,714 UBS Securities Australia Ltd Buy 1,867 1,1,714 UBS Securities Aus	Jun-14	UBS Securities Australia Ltd	Buy	80,889	8,340	Ordinary
UBS Securities Australia Ltd Buy 6,282 644 UBS Securities Australia Ltd Buy 6,282 6,241 UBS Securities Australia Ltd Buy 73,985 6,241 UBS Securities Australia Ltd Sell 73,985 6,648 UBS Securities Australia Ltd Sell 73,983 6,6015 UBS Securities Australia Ltd Sell 71,884 7,882 UBS Securities Australia Ltd Enter Into Derivative (NA) 7,138 7,138 UBS Securities Australia Ltd Buy 1,710 1,697 UBS Securities Australia Ltd Buy 1,710 1,42 UBS Securities Australia Ltd Buy 1,710 1,42 UBS Securities Australia Ltd Buy 1,768 1,927 UBS Securities Australia Ltd Buy 1,087 1,142 UBS Securities Australia Ltd Buy 1,087 1,144 UBS Securities Australia Ltd Buy 1,087 1,148 UBS Securities Australia Ltd Buy 1,087 1,148 UBS Securities A	Jun-14	UBS Securities Australia Ltd	Buy	97,514	10,000	Ordinary
UBS Securities Australia Ltd Buy 6,521 6,241 UBS Securities Australia Ltd Buy 283,487 24,582 UBS Securities Australia Ltd Sell 15,985 (1,548) (1,548) UBS Securities Australia Ltd Sell 5,333 (1,548) (1,580) UBS Securities Australia Ltd Enter Into Derivative C NA (21,500) (21,500) UBS Securities Australia Ltd Buy 1,697 (1,548) (1,587) (1,587) UBS Wealth Management Australia Ltd Buy 1,697 (1,114) (1,104) (1,104) UBS Wealth Management Australia Ltd Buy 1,897 (1,587) (1,587) (1,587) UBS Wealth Management Australia Ltd Buy 1,897 (1,587) (1,587) (1,587) UBS Wealth Management Australia Ltd Buy 1,897 (1,587) (1,587) (1,587) UBS Securities Australia Ltd Buy 1,897 (1,587) (1,587) (1,587) UBS Securities Australia Ltd Buy 1,897 (1,587) (1,587) <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>6,262</td> <td>644</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buy	6,262	644	Ordinary
UBS Securities Australia Ltd Buy 23,334,87 (15,686) UBS Securities Australia Ltd Buy 278,304 28,824 UBS Securities Australia Ltd Sell 278,304 28,824 UBS Securities Australia Ltd Sell 5,333 (5,015) UBS Securities Australia Ltd Enter Into Derivative C NA (90,200) UBS Securities Australia Ltd Buy 1,708 1,761 UBS Wealth Management Australia Ltd Buy 1,708 1,250 UBS Wealth Management Australia Ltd Buy 1,708 1,678 UBS Securities Australia Ltd Buy 1,708 1,678 UBS Securities Australia Ltd Buy 1,873 1,974 UBS Securities Australia Ltd Buy 3,847 3,544 UBS Securities Australia Ltd Buy 3,673 1,971 UBS Securities Australia Ltd Buy 3,559 3,674 UBS Securities Australia Ltd Buy 3,593 1,481 UBS Securities Australia Ltd Buy 3,593 1,849	7 - L	UBS Securities Australia Ltd	Buy	60,521	6,241	Ordinary
UBS Securities Australia Ltd Sell 115,985 (15,646) UBS Securities Australia Ltd Sell 15,985 28,324 UBS Securities Australia Ltd Sell 5,333 (15,646) UBS Securities Australia Ltd Enter Into Derivative C NAA (21,500) UBS Securities Australia Ltd Enter Into Derivative C NAA 388,400 UBS Wealth Management Australia Ltd Buy 17,108 1,761 UBS Wealth Management Australia Ltd Buy 17,108 1,761 UBS Wealth Management Australia Ltd Buy 18,7385 16,142 UBS Securities Australia Ltd Buy 18,7385 16,142 UBS Securities Australia Ltd Buy 18,7385 16,142 UBS Securities Australia Ltd Buy 18,7385 11,114 UBS Securities Australia Ltd Buy 18,738 11,144 UBS Securities Australia Ltd Buy 5,675 18,881 UBS Securities Australia Ltd Buy 18,09 1,648 UBS Securities Australia Ltd Buy 5,673	Jun-14	UBS Securities Australia Ltd	Buy	238,487	24,582	Ordinary
UBS Securities Australia Ltd Buy 28.824 UBS Securities Australia Ltd Sell CF44,191 C5.333 (548) UBS Securities Australia Ltd Enter Into Derivative C NVA (21,500) (21,500	Jun-14	UBS Securities Australia Ltd	Sell	151,985	(15,646)	Ordinary
UBS Securities Australia Ltd Sell 5,4191 (56,015) UBS Securities Australia Ltd Enter Into Derivative (1,42) NA (50,000) UBS Securities Australia Ltd Enter Into Derivative (1,108) NA (50,000) UBS Securities Australia Ltd Buy 17,108 1,761 UBS Wealth Management Australia Ltd Buy 17,108 1,761 UBS Wealth Management Australia Ltd Buy 182,722 2,957 UBS Wealth Management Australia Ltd Buy 183,078 19,290 UBS Securities Australia Ltd Buy 183,078 19,290 UBS Securities Australia Ltd Buy 183,078 19,290 UBS Securities Australia Ltd Buy 34,847 35,74 UBS Securities Australia Ltd Buy 36,75 567 UBS Securities Australia Ltd Buy 5,675 567 UBS Securities Australia Ltd Buy 5,675 567 UBS Securities Australia Ltd Buy 5,675 567 UBS Securities Australia Ltd Buy 1,3,67 1	Jun-14	UBS Securities Australia Ltd	Buy	278,304	28,824	Ordinary
UBS Securities Australia Ltd Sell (548) UBS Securities Australia Ltd Enter Into Derivative (1,500) NAA (50,200) UBS Securities Australia Ltd Enter Into Derivative (1,7,08 (21,500) (21,500) UBS Wealth Management Australia Ltd Buy 1,7,08 1,561 UBS Wealth Management Australia Ltd Buy 1,7,08 1,61,42 UBS Securities Australia Ltd Buy 1,57,385 16,142 UBS Securities Australia Ltd Buy 1,57,385 16,142 UBS Securities Australia Ltd Buy 1,57,385 16,142 UBS Securities Australia Ltd Buy 1,69 1,114 UBS Securities Australia Ltd Buy 1,69 1,114 UBS Securities Australia Ltd Buy 1,69 1,114 UBS Securities Australia Ltd Buy 1,69 1,164 UBS Securities Australia Ltd Buy 5,67 5,87 UBS Securities Australia Ltd Buy 1,6,90 1,1,14 UBS Securities Australia Ltd Buy 3,597 3,891	Jun-14	UBS Securities Australia Ltd	Sell	544,191	(56,015)	Ordinary
UBS Securities Australia Ltd Enter Into Derivative (10.50) NAA (90,200) UBS Securities Australia Ltd Enter Into Derivative (10.5486) 1,6486 1,697 UBS Wealth Management Australia Ltd Buy 17,108 1,697 UBS Wealth Management Australia Ltd Buy 28,127 2,957 UBS Wealth Management Australia Ltd Buy 28,127 2,957 UBS Securities Australia Ltd Buy 28,127 2,957 UBS Securities Australia Ltd Buy 36,471 1,114 UBS Securities Australia Ltd Buy 36,471 41,997 UBS Securities Australia Ltd Buy 36,575 1,687 UBS Securities Australia Ltd Buy 36,575 1,687 UBS Securities Australia Ltd Buy 6,591 6,591 6,687 UBS Securities Australia Ltd Buy 6,591 1,684 1,684 UBS Securities Australia Ltd Buy 6,591 1,684 1,684 UBS Securities Australia Ltd Buy 6,591 1,684 1,684 <	Jun-14	UBS Securities Australia Ltd	Sell	5,333	(548)	Ordinary
UBS Securities Australia Ltd Enter Into Derivative NAM (21,500) UBS Securities Australia Ltd Buy 1,708 1,761 UBS Wealth Management Australia Ltd Buy 28,727 2,937 UBS Wealth Management Australia Ltd Buy 28,727 2,937 UBS Securities Australia Ltd Buy 28,727 2,937 UBS Securities Australia Ltd Buy 18,787 19,290 UBS Securities Australia Ltd Buy 34,847 3,574 UBS Securities Australia Ltd Buy 34,847 3,574 UBS Securities Australia Ltd Buy 36,55 567 UBS Securities Australia Ltd Buy 3,59 367 UBS Securities Australia Ltd Buy 1,3,60 1,64 UBS Securities Australia Ltd Buy 3,59 1,64 UBS Securities Aust	Jun-14	UBS Securities Australia Ltd	Enter Into Derivative C	NA	(90,200)	Options
UBS Securities Australia Ltd Enter Into Derivative INA 388,400 UBS Wealth Management Australia Ltd Buy 17,108 1,561 UBS Wealth Management Australia Ltd Buy 17,108 1,561 UBS Securities Australia Ltd Buy 18,7385 16,142 UBS Securities Australia Ltd Buy 18,7385 16,142 UBS Securities Australia Ltd Buy 18,7385 16,142 UBS Securities Australia Ltd Buy 18,7385 11,114 UBS Securities Australia Ltd Buy 18,25 11,114 UBS Securities Australia Ltd Buy 10,85,50 11,082 UBS Securities Australia Ltd Buy 10,80,50 11,082 UBS Securities Australia Ltd Buy 10,80,50 11,082 UBS Securities Australia Ltd Buy 10,80,50 11,144 UBS Securities Australia Ltd Buy 10,80,50 11,144 UBS Securities Australia Ltd Buy 92,118 9,448 UBS Securities Australia Ltd Buy 92,118 9,488 <	Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(21,500)	Options
UBS Wealth Management Australia Ltd Buy 16,486 1,697 UBS Wealth Management Australia Ltd Buy 12,708 1,561 UBS Wealth Management Australia Ltd Buy 28,727 2,957 UBS Securities Australia Ltd Buy 186,078 19,290 UBS Securities Australia Ltd Buy 965 99 UBS Securities Australia Ltd Buy 36,471 41,997 UBS Securities Australia Ltd Buy 409,471 41,997 UBS Securities Australia Ltd Buy 409,471 41,997 UBS Securities Australia Ltd Buy 6,591 6,68 UBS Securities Australia Ltd Buy 6,591 6,591 UBS Securities Australia Ltd Buy 6,591 10,677 UBS Securities Australia Ltd Buy 6,591 10,677 UBS Securities Australia Ltd Buy 1,3,49 1,641 UBS Securities Australia Ltd Buy 2,718 9,48 UBS Securities Australia Ltd Buy 2,245 2,595 UBS Securiti	Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	388,400	Options
UBS Wealth Management Australia Ltd Buy 1,761 UBS Wealth Management Australia Ltd Buy 28,722 2,927 UBS Securities Australia Ltd Buy 18,017 19,290 UBS Securities Australia Ltd Buy 1,614 2,924 UBS Securities Australia Ltd Buy 34,847 3,574 UBS Securities Australia Ltd Buy 34,847 3,574 UBS Securities Australia Ltd Buy 409,362 11,114 UBS Securities Australia Ltd Buy 409,371 41,997 UBS Securities Australia Ltd Buy 5,573 578 UBS Securities Australia Ltd Buy 5,573 578 UBS Securities Australia Ltd Buy 3,529 3,671 UBS Securities Australia Ltd Buy 3,529 3,671 UBS Securities Australia Ltd Buy 3,529 3,671 UBS Securities Australia Ltd Buy 3,539 1,68 UBS Securities Australia Ltd Buy 3,539 1,68 UBS Securities Australia Ltd B	Jun-14	UBS Wealth Management Australia Ltd	Buy	16,486	1,697	Ordinary
UBS Securities Australia Ltd Buy 2.957 UBS Securities Australia Ltd Buy 189.078 19.200 UBS Securities Australia Ltd Buy 15,1385 16,142 99 UBS Securities Australia Ltd Buy 3,574 10 11.14 UBS Securities Australia Ltd Buy 3,675 11.114 UBS Securities Australia Ltd Buy 0,675 11.082 UBS Securities Australia Ltd Buy 0,675 1,082 UBS Securities Australia Ltd Buy 1,082 1,082 UBS Securities Australia Ltd Buy 3,529 3,831 UBS Securities Australia Ltd Buy 3,532 3,881 UBS Securities Australia Ltd Buy 3,537 3,891 UBS Securities Australia Ltd Buy 3,537 3,637 UBS Securities Aus	Jun-14	ustralia	Buy	17,108	1,761	Ordinary
UBS Securities Australia Ltd Buy 188,078 19,290 UBS Securities Australia Ltd Buy 157,385 19,290 UBS Securities Australia Ltd Buy 36,47 3,574 UBS Securities Australia Ltd Buy 36,72 11,114 UBS Securities Australia Ltd Buy 409,471 41,997 UBS Securities Australia Ltd Buy 6,591 41,997 UBS Securities Australia Ltd Buy 6,591 6,591 UBS Securities Australia Ltd Buy 82,118 9,448 UBS Securities Australia Ltd Buy 82,118 9,448 UBS Securities Australia Ltd Buy 11,349 1,164 UBS Securities Australia Ltd Buy 37,937 3,891 UBS Securities Australia Ltd Buy 37,937 1,64 UBS Securities Australia Ltd Buy 2,785 5,65 UBS Securities Australia Ltd Buy 2,24,50 2,53 UBS Securities Australia Ltd Buy 2,24,50 2,53 UBS Securities Australia Lt	Jun-14	UBS Wealth Management Australia Ltd	Buy	727,82	2,957	Ordinary
UBS Securities Australia Ltd Buy 15,385 16,142 UBS Securities Australia Ltd Buy 94 99 UBS Securities Australia Ltd Buy 34,847 3,574 UBS Securities Australia Ltd Buy 108,362 11,114 UBS Securities Australia Ltd Buy 409,471 41,997 UBS Securities Australia Ltd Buy 6,591 675 UBS Securities Australia Ltd Buy 6,591 6,76 UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 3,591 6,781 UBS Securities Australia Ltd Buy 3,593 3,891 UBS Securities Australia Ltd Buy 3,593 3,891 UBS Securities Australia Ltd Buy 3,793 1,840 UBS Securities Australia Ltd Buy 5,118 9,448 UBS Securities Australia Ltd Buy 2,245 2,500 UBS Securities Australia Ltd Buy 2,467 2,667 UBS Securities Australia Ltd <td< td=""><td>Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>188,078</td><td>19,290</td><td>Ordinary</td></td<>	Jun-14	UBS Securities Australia Ltd	Buy	188,078	19,290	Ordinary
UBS Securities Australia Ltd Buy 965 99 UBS Securities Australia Ltd Buy 34,847 3,724 UBS Securities Australia Ltd Buy 108,362 11,114 UBS Securities Australia Ltd Buy 40,947 11,082 UBS Securities Australia Ltd Buy 5,675 582 UBS Securities Australia Ltd Buy 6,591 6,591 6,591 UBS Securities Australia Ltd Buy 10,505 11,082 10,571 UBS Securities Australia Ltd Buy 3,529 3,891 10,571 UBS Securities Australia Ltd Buy 3,797 3,891 1,164 UBS Securities Australia Ltd Buy 3,797 3,891 1,164 UBS Securities Australia Ltd Buy 3,517 3,891 1,164 UBS Securities Australia Ltd Buy 3,517 3,891 1,164 UBS Securities Australia Ltd Buy 3,517 1,544 1,544 UBS Securities Australia Ltd Buy 2,467 2,513 1,544 </td <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>157,385</td> <td>16,142</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buy	157,385	16,142	Ordinary
UBS Securities Australia Ltd Buy 34,847 3,574 UBS Securities Australia Ltd Buy 10 11,114 UBS Securities Australia Ltd Buy 409,471 41,997 UBS Securities Australia Ltd Buy 409,471 41,997 UBS Securities Australia Ltd Buy 6,591 676 UBS Securities Australia Ltd Buy 6,591 676 UBS Securities Australia Ltd Buy 18,40,99 18,81 UBS Securities Australia Ltd Buy 18,40,99 18,81 UBS Securities Australia Ltd Buy 22,118 9,448 UBS Securities Australia Ltd Buy 37,937 11,64 UBS Securities Australia Ltd Buy 11,349 1,164 UBS Securities Australia Ltd Buy 22,118 1,267 UBS Securities Australia Ltd Buy 22,457 13,000 UBS Securities Australia Ltd Buy 22,457 13,000 UBS Securities Australia Ltd Buy 22,457 2,467 2,677 UBS	Jun-14	UBS Securities Australia Ltd	Buy	965	66	Ordinary
UBS Securities Australia Ltd Buy 10 UBS Securities Australia Ltd Buy 108,362 11,114 UBS Securities Australia Ltd Buy 5675 582 UBS Securities Australia Ltd Buy 5675 582 UBS Securities Australia Ltd Buy 5,575 587 UBS Securities Australia Ltd Buy 6,591 6,76 UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 35,937 10,881 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 55,136 1,64 UBS Securities Australia Ltd Buy 2,467 10,976 UBS Securities Australia Ltd Buy 2,467 10,976 UBS Securities Australia Ltd Buy 2,467 20,000 UBS Securities Australia Ltd Buy 2,	Jun-14	UBS Securities Australia Ltd	Buy	34,847	3,574	Ordinary
UBS Securities Australia Ltd Buy 108 362 11 114 UBS Securities Australia Ltd Buy 409 471 41,997 UBS Securities Australia Ltd Buy 56.75 11,082 UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 37,937 10,571 UBS Securities Australia Ltd Buy 37,937 3,831 UBS Securities Australia Ltd Buy 37,937 3,831 UBS Securities Australia Ltd Buy 37,178 9,448 UBS Securities Australia Ltd Buy 37,136 1,164 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 55,136 1,647 UBS Securities Australia Ltd Buy 2,245,500 230,000 UBS Securities Australia Ltd Buy 2,245,500 13,000 UBS Securities Australia Ltd Buy 2,245,500 13,000 UBS Securities A	Jun-14	UBS Securities Australia Ltd	Виу	10	-	Ordinary
UBS Securities Australia Ltd Buy 409,471 41,997 UBS Securities Australia Ltd Buy 108,050 11,002 UBS Securities Australia Ltd Buy 6,591 6,591 6,675 UBS Securities Australia Ltd Buy 6,591 6,591 6,676 365 UBS Securities Australia Ltd Buy 6,591 6,691 <t< td=""><td>Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>108,362</td><td>11,114</td><td>Ordinary</td></t<>	Jun-14	UBS Securities Australia Ltd	Buy	108,362	11,114	Ordinary
UBS Securities Australia Ltd Buy 5675 582 UBS Securities Australia Ltd Buy 1080.550 11,082 UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 11,349 1,64 UBS Securities Australia Ltd Buy 55,136 1,64 UBS Securities Australia Ltd Buy 55,136 1,64 UBS Securities Australia Ltd Buy 2,467 20,000 UBS Securities Australia Ltd Buy 2,124,500 1,000 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buy	409,471	41,997	Ordinary
UBS securities Australia Ltd Buy 108 CSC UBS Securities Australia Ltd Buy 6,591 667 UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 103,067 10,571 UBS Securities Australia Ltd Buy 22,118 9,448 UBS Securities Australia Ltd Buy 37,937 3,831 UBS Securities Australia Ltd Buy 37,937 3,831 UBS Securities Australia Ltd Buy 11,349 1,464 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 2,245,500 230,000 UBS Securities Australia Ltd Buy 2,245,500 230,000 UBS Securities Australia Ltd Buy 2,245,500 13,000 UBS Securities Australia Ltd Buy 2,245,500 13,000 UBS Securities Australia Ltd Buy 2,125,500 13,000 UBS Securities Australia Ltd <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>5,675</td> <td>285</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buy	5,675	285	Ordinary
UBS Securities Australia Ltd Buy 6,591 676 UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 184,090 18,811 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 37,937 3,881 UBS Securities Australia Ltd Buy 37,937 3,881 UBS Securities Australia Ltd Buy 11,349 1,164 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 22,425,00 28,00 UBS Securities Australia Ltd Buy 22,425,00 230,00 UBS Securities Australia Ltd Buy 22,425,00 13,000 UBS Securities Australia Ltd Buy 22,425,00 230,000 UBS Securities Australia Ltd Buy 22,425,00 230,000 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securit	Jun-14	UBS Securities Australia Ltd	Buy	108,050	11,082	Ordinary
UBS Securities Australia Ltd Buy 3,559 365 UBS Securities Australia Ltd Buy 18,4090 18,881 UBS Securities Australia Ltd Buy 92,118 9,448 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 32,118 9,448 UBS Securities Australia Ltd Buy 13,528 1,64 UBS Securities Australia Ltd Buy 55,136 1,64 UBS Securities Australia Ltd Buy 55,136 5,65 UBS Securities Australia Ltd Buy 2,467 20,000 UBS Securities Australia Ltd Buy 2,124 2,146 UBS Securities Australia Ltd Buy 2,124 2,146 UBS Securities Australia Ltd Buy 2,124 2,184 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buy	6,591	9/9	Ordinary
UBS Securities Australia Ltd Buy 184,090 18,881 UBS Securities Australia Ltd Buy 10,577 10,577 UBS Securities Australia Ltd Buy 37,937 3,891 UBS Securities Australia Ltd Buy 37,937 3,891 UBS Securities Australia Ltd Buy 11,1549 1,164 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 25,136 5,655 UBS Securities Australia Ltd Buy 2,245,00 230,000 UBS Securities Australia Ltd Buy 2,124,500 13,000 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 1,264 UBS	Jun-14	UBS Securities Australia Ltd	Buy	3,559	365	Ordinary
UBS Securities Australia Ltd Buy 103,067 10,571 UBS Securities Australia Ltd Buy 22,118 9,448 UBS Securities Australia Ltd Buy 37,937 3,831 UBS Securities Australia Ltd Buy 11,349 1,164 UBS Securities Australia Ltd Buy 653 68 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 10,206 10,976 UBS Securities Australia Ltd Buy 2,24,57 28,000 UBS Securities Australia Ltd Buy 2,24,57 13,000 UBS Securities Australia Ltd Buy 2,24,57 13,000 UBS Securities Australia Ltd Buy 2,24,50 13,000 UBS Securities Australia Ltd Buy 2,1,54 2,1,30 UBS Securities Australia Ltd Buy 2,1,34 2,1,84 UBS Securities Australia Ltd Buy 2,1,24 2,1,84 UBS Securities Australia Ltd Buy 2,1,24 2,184 UBS Securities Austr	Jun-14	UBS Securities Australia Ltd	Buv	184 090	18 881	Ordinary
UBS Securities Australia Ltd Buy 92,118 9,448 UBS Securities Australia Ltd Buy 37,937 3,891 UBS Securities Australia Ltd Buy 11,349 1,144 UBS Securities Australia Ltd Buy 13,528 13,490 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 2,245,500 230,000 UBS Securities Australia Ltd Buy 2,245,500 13,000 UBS Securities Australia Ltd Buy 7,245,500 13,000 UBS Securities Australia Ltd Buy 7,245,500 13,000 UBS Securities Australia Ltd Buy 7,129,4 2,184 UBS Securities Australia Ltd Buy 2,124 2,184 UBS Securities Australia Ltd Buy 2,124 2,184 UBS Securities Australia Ltd Buy 2,124 2,184 UBS Securities Australia Ltd Buy 1,254 1,124 UBS Securitie	Jun-14	UBS Securities Australia Ltd	Buv	103.067	10 571	Ordinary
UBS Securities Australia Ltd Buy 37,937 3,891 UBS Securities Australia Ltd Buy 11,349 1,164 UBS Securities Australia Ltd Buy 663 68 UBS Securities Australia Ltd Buy 55,136 1,655 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 2,242,500 230,000 UBS Securities Australia Ltd Buy 2,242,500 230,000 UBS Securities Australia Ltd Buy 2,242,500 13,000 UBS Securities Australia Ltd Buy 2,242,500 13,000 UBS Securities Australia Ltd Buy 2,1410 760 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 12,324 1,264 UBS Securities Australia Ltd Buy 1,254 1,264 UBS Securities Australia Ltd Buy 1,264 1,264 UBS Securities Austr	Jun-14	UBS Securities Australia Ltd	Buv	92.118	9.448	Ordinary
UBS Securities Australia Ltd Buy 11,329 1,020 UBS Securities Australia Ltd Buy 11,329 1,164 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 2,24,50 28,00 UBS Securities Australia Ltd Buy 2,24,50 28,00 UBS Securities Australia Ltd Buy 1,26,75 13,000 UBS Securities Australia Ltd Buy 7,410 760 UBS Securities Australia Ltd Buy 21,294 2,134 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 2,224 2,134 UBS Securities Australia Ltd Buy 2,1,294 2,1,84 UBS Securities Australia Ltd Buy 1,2,224 1,1,84 UBS Securities Australia Ltd Buy 1,2,24 1,264 UBS Securities Australia	lun-14	UBS Securities Australia Itd	Bliv	37 937	2 801	Ordingo
UBS Securities Australia Ltd Buy 18.56 UBS Securities Australia Ltd Buy 13.15.28 13.490 UBS Securities Australia Ltd Buy 55,136 5.655 UBS Securities Australia Ltd Buy 55,136 10,976 UBS Securities Australia Ltd Buy 2,245,500 230,000 UBS Securities Australia Ltd Buy 7,245,500 13,000 UBS Securities Australia Ltd Buy 7,245,500 13,000 UBS Securities Australia Ltd Buy 7,120 760 UBS Securities Australia Ltd Buy 21,234 2,134 UBS Securities Australia Ltd Buy 21,234 2,184 UBS Securities Australia Ltd Buy 21,234 2,184 UBS Securities Australia Ltd Buy 12,224 1,1264 UBS Securities Australia Ltd Buy 1,224 2,124 UBS Securities Australia Ltd Buy 1,224 1,1264 UBS Securities Australia Ltd Buy 1,224 1,1264 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Birv	11 349	1 164	Ordinary
UBS Securities Australia Ltd Buy 131,528 13,490 UBS Securities Australia Ltd Buy 55,136 13,490 UBS Securities Australia Ltd Buy 2,467 10,976 UBS Securities Australia Ltd Buy 2,467 233 UBS Securities Australia Ltd Buy 2,242,500 230,000 UBS Securities Australia Ltd Buy 7,410 760 UBS Securities Australia Ltd Buy 7,110 760 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 12,324 1,264 UBS Securities Australia Ltd Buy 1,264 1,264 UBS Securities Australia Ltd Buy 1,264 1,264 UBS Securities Australia Ltd <td>Jun-14</td> <td>UBS Securities Australia I1d</td> <td>Biry</td> <td>599</td> <td>1 0</td> <td>Crainary</td>	Jun-14	UBS Securities Australia I1d	Biry	599	1 0	Crainary
UBS Securities Australia Ltd Buy 15,728 UBS Securities Australia Ltd Buy 55,136 5,655 UBS Securities Australia Ltd Buy 2,467 10,976 UBS Securities Australia Ltd Buy 2,242,500 230,000 UBS Securities Australia Ltd Buy 1,26,750 13,000 UBS Securities Australia Ltd Buy 7,410 760 UBS Securities Australia Ltd Buy 21,294 2,134 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 12,254 2,124 UBS Securities Australia Ltd Buy 1,254 1,264 UBS Securities Australia Ltd Buy 1,254 1,264 UBS Securities Australia Ltd Buy 1,254 1,264	lin-14	I IRS Societies Australia 14d	62.0	121 530	00 00	Ordinary
UBS Securities Australia Ltd Buy 107,015 0.976 UBS Securities Australia Ltd Buy 107,015 <	hn-14	LIBS Securities Australia 11d	200	35,151	15,450	Ordinary
UBS Securities Australia Ltd Buy 2,467 153 UBS Securities Australia Ltd Buy 2,242,500 230,000 UBS Securities Australia Ltd Buy 7,410 13,000 UBS Securities Australia Ltd Buy 7,410 7,60 UBS Securities Australia Ltd Buy 7,410 7,710 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 1,254 3,184 UBS Securities Australia Ltd Buy 1,234 1,234 UBS Securities Australia Ltd Buy 1,2324 1,234 UBS Securities Australia Ltd Buy 1,2324 1,234 UBS Securities Australia Ltd Buy 1,2324 1,234	114	I IRS Societies Australia Ltd	and a	20,150	250.01	Ordinary
UBS Securities Australia Ltd Buy 2,245,00 23,000 UBS Securities Australia Ltd Buy 2,245,60 230,000 UBS Securities Australia Ltd Buy 1,26,750 13,000 UBS Securities Australia Ltd Buy 7,410 761 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 12,244 374 UBS Securities Australia Ltd Buy 1,264 3,184 UBS Securities Australia Ltd Buy 1,264 1,264 UBS Securities Australia Ltd Buy 1,264 1,264 UBS Securities Australia Ltd Buy 1,264 1,264	17	1 BC Counties Australia Ltd	600	200,101	0/6/01	Ordinary
UBS Securities Australia Ltd Buy 2,20,000 UBS Securities Australia Ltd Buy 7,410 76,750 UBS Securities Australia Ltd Buy 7,410 760 UBS Securities Australia Ltd Buy 7,1294 2,1294 UBS Securities Australia Ltd Buy 7,1294 2,184 UBS Securities Australia Ltd Buy 7,2224 2,184 UBS Securities Australia Ltd Buy 1,224 1,264	17	IRS Societies Australia 14d	buy.	7075 676 6	730,000	Orginary
UBS Securities Australia Ltd Buy 7.20,730 1.000 UBS Securities Australia Ltd Buy 7.1294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 12,324 1,264 UBS Securities Australia Ltd Buy 12,324 1,264 UBS Securities Australia Ltd Buy 12,7862 1,314	17	I IBC Countries Australia I to	pro.	005,242,500	730,000	Orginary
UBS Securities Australia Ltd Buy 7,410 760 UBS Securities Australia Ltd Buy 1392,01 14,234 UBS Securities Australia Ltd Buy 21,294 2,184 UBS Securities Australia Ltd Buy 21,294 374 UBS Securities Australia Ltd Buy 12,324 1,264 UBS Securities Australia Ltd Buy 12,324 1,264 UBS Securities Australia Ltd Buy 12,786 1,314	11-17	Upo Commission Augustiania Liu	(in a	06/97	13,000	Ordinary
UBS Securities Australia Ltd Buy 194,201 14,277 UBS Securities Australia Ltd Buy 2,124 3,647 37.4 UBS Securities Australia Ltd Buy 3,647 37.4 184 UBS Securities Australia Ltd Buy 1,2,324 1,264 1,264 UBS Securities Australia Ltd Buy 12,765 13,114	4	UBS Securities Australia Ltd	any	0,410	09/	Ordinary
UBS Securities Australia Ltd Buy 2.184 2,184 UBS Securities Australia Ltd Buy 3,647 374 UBS Securities Australia Ltd Buy 12,324 1,264 UBS Securities Australia Ltd Buy 12,786 13,114	Un-14	UBS Securities Australia Ltd	Ruy	107'651	14,277	Ordinary
UBS Securities Australia Ltd Buy 3.56.7 37.4 UBS Securities Australia Ltd Buy 12,32.4 1,564 UBS Securities Australia Ltd Buy 12,32.4 1,564 UBS Securities Australia Ltd Buy 12,76.2 13,114	un-14	UBS Securities Australia Ltd	Buy	21,294	2,184	Ordinary
UBS Securities Australia Ltd Buy 12,324 1,264 UBS Securities Australia Ltd Buy 127,862 13,114	41-I	UBS Securities Australia Ltd	Buy	3,647	374	Ordinary
UBS Securities Australia Ltd Buy 127,862 13,114	nn-14	UBS Securities Australia Ltd	Buy	12,324	1.264	Ordinan
	77					Cidillary

				707	1.7	Ordinary
UBS Securities Australia Ltd Buy 6.27 6.27 UBS Securities Australia Ltd Buy 1.251 1.93 UBS Securities Australia Ltd Buy 1.251 1.231 UBS Securities Australia Ltd Buy 1.255 2.246 UBS Securities Australia Ltd Buy 2.259 2.246 UBS Securities Australia Ltd Buy 2.659 2.76 UBS Securities Australia Ltd Buy 2.659 2.246 UBS Securities Australia Ltd Buy 2.659 2.670 UBS Securities Australia Ltd Buy 2.629 2.670 UBS Securities Australia Ltd Buy 2.750 2.75 UBS Securities Australia Ltd Buy 2.750 2.75 UBS Securities Australia Ltd Buy 2.750 2.75 UBS Securities Australia Ltd Buy<	1-Jun-14	UBS Securities Australia Ltd	soo.	050.0	15.	
USS Securities Australia Ltd Buy 1,2,201 1,515	lin-14	UBS Securities Australia 11d	Riny	6.024	234	Ordinary
USS Securities Avaitable Ltd Boby 1,2,251 1,2,63 USS Securities Avaitable Ltd Boby 1,2,251 1,2,63 USS Securities Avaitable Ltd Boy 1,2,70 2,240 USS Securities Avaitable Ltd Boy 2,2,70 2,240 USS Securities Avaitable Ltd Boy 2,246 1,19 USS Securities Avaitable Ltd Boy 2,240 2,240 USS Securities Avaitable Ltd Boy 2,270 2,240 USS Securities Avaitable Ltd Boy 2,240 2,240 USS Securities Avaitable Ltd Boy 1,756 1,590 USS Securities Avaitable Ltd Boy 1,756 1,590 USS Securities Avaitable Ltd Boy 1,756 1,500 USS Securities Avaitable Ltd Boy 1,756 1,500 USS Securities Avaitable Ltd Boy 1,756 1,700 USS Securities Avaitable Ltd Boy 2,500 1,700 USS Securities Avaitable Ltd Boy 2,500 1,700 USS Securities Avaitable Ltd	kın-14	I IBS Securities Australia 11d	Bury	184	100	Ordinary
USS Securities Australia Lud Buy 2,220 7.24 USS Securities Australia Lud Buy 2,220 2,240 USS Securities Australia Lud Buy 2,270 2,240 USS Securities Australia Lud Buy 2,520 2,341 USS Securities Australia Lud Buy 2,520 2,342 USS Securities Australia Lud Buy 2,520 2,342 USS Securities Australia Lud Buy 2,520 2,342 USS Securities Australia Lud Buy 2,520 2,501 USS Securities Australia Lud Buy 1,250 1,529 USS Securities Australia Lud Buy 1,250 1,529 USS Securities Australia Lud Buy 2,510 2,000 USS Securities Australia Lud Buy 2,510 3,000 USS Securities Australia Lud Buy 2,520 2,510 USS Securities Australia Lud Buy 2,520 2,510 USS Securities Australia Lud Buy 2,520 2,520 USS Securities Australia Lud <t< td=""><td>11-14</td><td>LIRS Securities Australia 14d</td><td>200</td><td>12.251</td><td>1 363</td><td>Ordinary</td></t<>	11-14	LIRS Securities Australia 14d	200	12.251	1 363	Ordinary
USE Securities Australia LID Buy 1,2,20 2,24 USE Securities Australia LID Buy 1,2,40 2,240 USE Securities Australia LID Buy 1,2,70 2,246 USE Securities Australia LID Buy 1,2,70 2,246 USE Securities Australia LID Buy 2,270 2,242 USE Securities Australia LID Buy 2,270 2,242 USE Securities Australia LID Buy 1,2,50 1,31 USE Securities Australia LID Buy 1,2,50 1,32 USE Securities Australia LID Buy 1,2,50 1,2,20 USE Securities Australia LID Buy 1,2,23 1,2,60 USE Securities Australia LID Buy 1,2,20 1,2,20 USE Securities Australia L	11/4	I IRS Securities Australia 14d	Biny	026.0	700	Ordings
URS Securities Australia Ltd Buy 2,270 LOSS Securities Australia Ltd Buy 2,676 2,486 LOSS Securities Australia Ltd Buy 2,670 2,486 LOSS Securities Australia Ltd Buy 2,570 2,571 LOSS Securities Australia Ltd Buy 2,570 2,371 LOSS Securities Australia Ltd Buy 2,570 2,371 LOSS Securities Australia Ltd Buy 1,136 1,137 LOSS Securities Australia Ltd Buy 1,136 1,139 LOSS Securities Australia Ltd Buy 1,22,239 1,260 LOSS Securities Australia Ltd Buy 1,22,239 1,260 LOSS Securities Australia Ltd Buy 2,514 860 LOSS Securities Australia Ltd Buy 2,529 1,700 LOSS Securities Australia Ltd Buy 2,539 1,700 LOSS Securities Australia Ltd Buy 2,539 1,700 LOSS Securities Australia Ltd Buy 2,539 1,700 LOSS Securities Australia Ltd B	11-14	I BS Securities Australia Ltd	Bery	19 400	0000	Ordinary
UBS Securities Australia Ltd Buy 2,656 2,546 UBS Securities Australia Ltd Buy 2,659 2,546 UBS Securities Australia Ltd Buy 2,559 2,422 UBS Securities Australia Ltd Buy 2,570 2,422 UBS Securities Australia Ltd Buy 2,570 1,531 UBS Securities Australia Ltd Buy 2,570 1,531 UBS Securities Australia Ltd Buy 2,570 1,531 UBS Securities Australia Ltd Buy 1,756 1,831 UBS Securities Australia Ltd Buy 1,524 1,801 UBS Securities Australia Ltd <	12-14	LIBS Securities Australia Ltd	Bis	2 220	750	Ordinary
URS Securities Australia Ltd Buy 1,554 1,19 UBS Securities Australia Ltd Buy 2,270 2,34 UBS Securities Australia Ltd Buy 2,270 2,34 UBS Securities Australia Ltd Buy 1,056 1,056 UBS Securities Australia Ltd Buy 1,126 1,33 UBS Securities Australia Ltd Buy 1,262 1,36 UBS Securities Australia Ltd Buy 1,262 1,36 UBS Securities Australia Ltd Buy 1,22,39 1,560 UBS Securities Australia Ltd Buy 4,36 4,48 UBS Securities Australia Ltd Buy 2,310 2,910 UBS Securities Australia Ltd Buy 2,910 2,910 UBS Securities Australia Ltd Buy	lısn-14	URS Securities Australia Ltd	Bir	20 696	2 546	Ordinary
LBS Securities Australia Ltd Buy 26.59 2.74 LBS Securities Australia Ltd Buy 26.59 2.74 LBS Securities Australia Ltd Buy 1,355 1,17 LBS Securities Australia Ltd Buy 1,250 1,250 LBS Securities Australia Ltd Buy 7,954 823 LBS Securities Australia Ltd Buy 7,954 823 LBS Securities Australia Ltd Buy 7,250 1,250 LBS Securities Australia Ltd Buy 1,260 1,250 LBS Securities Australia Ltd Buy 4,386 540 LBS Securities Australia Ltd Buy 2,316 30 LBS Securities Australia Ltd Buy 2,238 54 LBS Securities Australia Ltd Buy 2,238 54 LBS Securities Australia Ltd Buy 2,238 54 LBS Securities Australia Ltd Buy 2,236 1,704 LBS Securities Australia Ltd Buy 2,238 54 LBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia I td	Bir	1 154	011	Ordinary
UBS Securities Australia Ltd Buy 26,597 2,742 UBS Securities Australia Ltd Buy 1,356 1,568 UBS Securities Australia Ltd Buy 1,356 1,568 UBS Securities Australia Ltd Buy 1,270 2,74 UBS Securities Australia Ltd Buy 1,269 1,598 UBS Securities Australia Ltd Buy 1,2239 1,260 UBS Securities Australia Ltd Buy 1,2239 1,260 UBS Securities Australia Ltd Buy 4,366 4,48 UBS Securities Australia Ltd Buy 4,366 4,48 UBS Securities Australia Ltd Buy 2,310 3,00 UBS Securities Australia Ltd Buy 2,310 3,00 UBS Securities Australia Ltd Buy 2,310 3,00 UBS Securities Australia Ltd Buy 2,329 1,704 UBS Securities Australia Ltd Buy 2,329 1,704 UBS Securities Australia Ltd Buy 1,524 1,704 UBS Securities Australia Ltd	lun-14	I IBS Securities Australia Ltd	Bus	051	g	Ordingo
UBS Securities Australia Ltd Buy 2,270 2,33 UBS Securities Australia Ltd Buy 1,135 1,135 UBS Securities Australia Ltd Buy 1,756 1,831 UBS Securities Australia Ltd Buy 7,756 1,831 UBS Securities Australia Ltd Buy 7,554 820 UBS Securities Australia Ltd Buy 1,240 2,00 UBS Securities Australia Ltd Buy 1,240 2,00 UBS Securities Australia Ltd Buy 1,240 2,00 UBS Securities Australia Ltd Buy 2,910 3,00 UBS Securities Australia Ltd Buy 4,346 4,48 UBS Securities Australia Ltd Buy 4,281 1,79 UBS Securities Australia Ltd Buy 1,524 1,70 UBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia I td	Blay	26.597	27.7	Ordinan
LIBS Securities Australia Ltd Bluy 1,135 1171 LIBS Securities Australia Ltd Bluy 1,736 1,688 LIBS Securities Australia Ltd Bluy 1,756 1,331 LIBS Securities Australia Ltd Bluy 7,546 820 LIBS Securities Australia Ltd Bluy 7,546 820 LIBS Securities Australia Ltd Bluy 7,546 820 LIBS Securities Australia Ltd Bluy 7,547 4,481 LIBS Securities Australia Ltd Bluy 2,310 3,00 LIBS Securities Australia Ltd Bluy 2,310 3,00 LIBS Securities Australia Ltd Bluy 2,310 3,00 LIBS Securities Australia Ltd Bluy 5,310 3,00 LIBS Securities Australia Ltd Bluy 5,320 1,301 LIBS Securities Australia Ltd Bluy 5,320 1,301 LIBS Securities Australia Ltd Bluy 5,320 1,301 LIBS Securities Australia Ltd Bluy 1,322 1,301 LIBS Securities Australi	lun-14	UBS Securities Australia Ltd	Bis	2 270	734	Ordinary
UBS Securities Australia Ltd Buy 1,786 1,831 UBS Securities Australia Ltd Buy 1,784 1,831 UBS Securities Australia Ltd Buy 1,784 1,831 UBS Securities Australia Ltd Buy 1,2600 1,290 UBS Securities Australia Ltd Buy 1,249 1,260 UBS Securities Australia Ltd Buy 2,910 3,00 UBS Securities Australia Ltd Buy 2,928 5,74 UBS Securities Australia Ltd Buy 5,28 5,74 UBS Securities Australia Ltd Buy 5,28 5,74 UBS Securities Australia Ltd Buy 5,28 5,74 UBS Securities Australia Ltd Buy	14 In-14	UBS Securities Australia Ltd	Birv	1 135	117	Ordinan
UBS Securities Australia Ltd Buy 17,756 1,831 UBS Securities Australia Ltd Buy 7,946 820 UBS Securities Australia Ltd Buy 7,946 820 UBS Securities Australia Ltd Buy 7,946 1,200 UBS Securities Australia Ltd Buy 2,910 1,200 UBS Securities Australia Ltd Buy 2,910 2,910 UBS Securities Australia Ltd Buy 2,910 2,910 UBS Securities Australia Ltd Buy 2,910 3,00 UBS Securities Australia Ltd Buy 2,568 4,48 UBS Securities Australia Ltd Buy 5,568 3,54 UBS Securities Australia Ltd Buy 5,280 1,704 UBS Securities Australia Ltd Buy 5,280 1,704 UBS Securities Australia Ltd Buy <td>hn-14</td> <td>UBS Securities Australia I td</td> <td>Bliv</td> <td>10 360</td> <td>1 068</td> <td>Ordinary</td>	hn-14	UBS Securities Australia I td	Bliv	10 360	1 068	Ordinary
UBS Securities Australia Ltd Buy 1,26 1,29 UBS Securities Australia Ltd Buy 1,260 1,290 UBS Securities Australia Ltd Buy 1,260 1,290 UBS Securities Australia Ltd Buy 4,481 462 UBS Securities Australia Ltd Buy 4,281 462 UBS Securities Australia Ltd Buy 4,381 481 UBS Securities Australia Ltd Buy 5,238 540 UBS Securities Australia Ltd Buy 5,238 541 UBS Securities Australia Ltd Buy 5,281 1,213 UBS Securities Australia Ltd Buy 5,284 1,213 UBS Securities Australia Ltd Buy 5,284 1,704 UBS Securities Australia Ltd Buy 5,284 1,704 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Bary	17.761	1 831	Ordinary
UBS Securities Australia Ltd Bby 19,400 2,000 UBS Securities Australia Ltd Buy 1,9400 2,000 UBS Securities Australia Ltd Buy 1,269 1,000 UBS Securities Australia Ltd Buy 1,22,39 1,000 UBS Securities Australia Ltd Buy 2,910 300 UBS Securities Australia Ltd Buy 1,1543 1,190 UBS Securities Australia Ltd Buy 1,265 1,704 UBS Securities Australia Ltd Buy 1,265 1,704 UBS Securities Australia Ltd Buy 1,369 1,704 UBS Securities Australia Ltd Buy 1,369 1,704 UBS Securities Australia Ltd Buy 1,369 1,704 UBS Securities Australia Ltd Bu	Jun-14	UBS Securities Australia Ltd	Birv	126	13	Ordinary
UBS Securities Australia Ltd Buy 1,2,600 2,000 UBS Securities Australia Ltd Buy 1,2,600 1,2,93 UBS Securities Australia Ltd Buy 4,481 4,62 UBS Securities Australia Ltd Buy 2,910 4,00 UBS Securities Australia Ltd Buy 2,917 6,00 UBS Securities Australia Ltd Buy 4,65 7,00 UBS Securities Australia Ltd Buy 5,568 5,74 UBS Securities Australia Ltd Buy 1,734 1,704 UBS Securities Australia Ltd Buy 1,529 1,505 UBS Securities Australia Ltd Buy 1,530 1,505 UBS Securities Australia Ltd Buy 1,530 1,505 UBS Securities Australia Ltd Bu	lun-14	UBS Securities Australia Ltd	Bly	7 954	620	Ordinary
UBS Securities Australia Ltd Buy 1,2,600 1,239 UBS Securities Australia Ltd Buy 1,2,600 1,239 UBS Securities Australia Ltd Buy 1,2,600 1,239 UBS Securities Australia Ltd Buy 2,910 300 UBS Securities Australia Ltd Buy 2,910 30 UBS Securities Australia Ltd Buy 2,910 30 UBS Securities Australia Ltd Buy 2,568 5,48 UBS Securities Australia Ltd Buy 1,532 1,704 UBS Securities Australia Ltd Buy 1,522 1,704 UBS Securities Australia Ltd Buy 1,529 1,504 UBS Securities Australia Ltd Buy 1,520 1,504 UBS Securities Australia Ltd Buy 1,520 1,504 UBS Securities Australia Ltd Buy <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy.</td> <td>19.400</td> <td>2,000</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buy.	19.400	2,000	Ordinary
UBS Securities Australia Ltd Buy 4,61 4,62 UBS Securities Australia Ltd Buy 1,22,339 1,2,62 UBS Securities Australia Ltd Buy 2,910 3,00 UBS Securities Australia Ltd Buy 2,526 5,27 84 UBS Securities Australia Ltd Buy 4,559 1,781 1,781 UBS Securities Australia Ltd Buy 4,559 1,764 1,855 UBS Securities Australia Ltd Buy 1,534 1,564 1,856 1,564 UBS Securities Australia Ltd Buy 1,529 1,504 1,664 1,664 1,664 1,664 1,664 1,664 1,664 1,664 1,664 1,664	Jun-14	UBS Securities Australia Ltd	Buv	12.600	1,299	Ordinary
UBS Securities Australia Ltd Buy 12,239 12,602 UBS Securities Australia Ltd Buy 2,910 300 UBS Securities Australia Ltd Buy 2,910 300 UBS Securities Australia Ltd Buy 4,381 4,481 UBS Securities Australia Ltd Buy 4,391 30 UBS Securities Australia Ltd Buy 4,381 4,98 UBS Securities Australia Ltd Buy 4,582 4,88 UBS Securities Australia Ltd Buy 1,754 4,88 UBS Securities Australia Ltd Buy 1,754 1,704 UBS Securities Australia Ltd Buy 1,734 1,704 UBS Securities Australia Ltd Buy 1,733 1,704 UBS Securities Australia Ltd Buy 1,734 1,704 UBS Securities Australia Ltd Buy 1,734 1,704 UBS Securities Australia Ltd Buy 1,734 1,704 UBS Securities Australia Ltd Buy 5,286 1,704 UBS Securities Australia Ltd Buy </td <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>4,481</td> <td>462</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buy	4,481	462	Ordinary
UBS Securities Australia Ltd Bby 2,910 300 UBS Securities Australia Ltd Buy 2,910 448 UBS Securities Australia Ltd Buy 5,288 448 UBS Securities Australia Ltd Buy 5,281 29 UBS Securities Australia Ltd Buy 2,291 30 UBS Securities Australia Ltd Buy 2,21 43 UBS Securities Australia Ltd Buy 2,28 22 UBS Securities Australia Ltd Buy 5,287 542 UBS Securities Australia Ltd Buy 5,287 542 UBS Securities Australia Ltd Buy 5,588 4,895 UBS Securities Australia Ltd Buy 1,786 1,787 UBS Securities Australia Ltd Buy 1,534 1,786 1,787 UBS Securities Australia Ltd Buy 1,532 1,766 1,862 UBS Securities Australia Ltd Buy 1,534 1,766 1,767 UBS Securities Australia Ltd Buy 1,529 1,766 1,766 </td <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>800</td> <td>122,239</td> <td>12.602</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	800	122,239	12.602	Ordinary
UBS Securities Australia Ltd Bby 4,346 448 UBS Securities Australia Ltd Buy 5,128 540 UBS Securities Australia Ltd Buy 2,910 2,91 UBS Securities Australia Ltd Buy 475 49 UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 6,527 8 UBS Securities Australia Ltd Buy 6,529 484 UBS Securities Australia Ltd Buy 1,734 1,787 UBS Securities Australia Ltd Buy 1,6229 1,704 UBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia Ltd	Buy	2.910	300	Ordinary
UBS Securities Australia Ltd Buy 5,238 540 UBS Securities Australia Ltd Buy 2,310 300 UBS Securities Australia Ltd Buy 4,75 49 UBS Securities Australia Ltd Buy 1,543 1,190 UBS Securities Australia Ltd Buy 1,543 1,190 UBS Securities Australia Ltd Buy 7,88 5,42 UBS Securities Australia Ltd Buy 7,88 5,42 UBS Securities Australia Ltd Buy 1,756 1,291 UBS Securities Australia Ltd Buy 1,523 1,764 UBS Securities Australia Ltd Buy 1,529 1,764 UBS Securities Australia Ltd Buy 1,530 1,766 UBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia Ltd	Buy	4,346	448	Ordinary
UBS Securities Australia Ltd Buy 2,910 300 UBS Securities Australia Ltd Buy 475 49 UBS Securities Australia Ltd Buy 281 29 UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 5,257 327 UBS Securities Australia Ltd Buy 1,766 1,213 UBS Securities Australia Ltd Buy 4,695 484 UBS Securities Australia Ltd Buy 1,766 1,704 UBS Securities Australia Ltd Buy 1,6529 1,704 UBS Securities Australia Ltd Buy 1,530 1,930 UBS Securities Australia Ltd Buy 1,530 1,064 UBS Securities Australia Ltd Buy 1,530 1,064 UBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia Ltd	Buy	5,238	540	Ordinary
UBS Securities Australia Ltd Buy 475 49 UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 5.257 54 UBS Securities Australia Ltd Buy 1,786 1,213 UBS Securities Australia Ltd Buy 1,786 1,213 UBS Securities Australia Ltd Buy 1,529 1,704 UBS Securities Australia Ltd Buy 1,539 1,504 UBS Securities Australia Ltd Buy 1,539 1,504 UBS Securities Australia Ltd Buy 1,539 1,504 UBS Securities Australia Ltd Buy 5,386 1,614 UBS Securities Australia Ltd Buy 1,338	lun-14	UBS Securities Australia Ltd	Buy	2,910	300	Ordinary
UBS Securities Australia Ltd Buy 281 29 UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 1,543 1,190 UBS Securities Australia Ltd Buy 1,543 1,190 UBS Securities Australia Ltd Buy 1,558 5,74 UBS Securities Australia Ltd Buy 1,7334 1,787 UBS Securities Australia Ltd Buy 1,7334 1,787 UBS Securities Australia Ltd Buy 1,5229 1,704 UBS Securities Australia Ltd Buy 1,5229 1,704 UBS Securities Australia Ltd Buy 1,5239 1,704 UBS Securities Australia Ltd Buy 1,5229 1,704 UBS Securities Australia Ltd Buy 1,529 1,505 UBS Securities Australia Ltd Buy 1,530 1,90 UBS Securities Australia Ltd Buy 1,530 1,064 UBS Securities Australia Ltd Buy 1,530 1,064 UBS Securities Australia Ltd Buy </td <td>lun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>475</td> <td>49</td> <td>Ordinary</td>	lun-14	UBS Securities Australia Ltd	Buy	475	49	Ordinary
UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 11,543 1,190 UBS Securities Australia Ltd Buy 5,257 542 UBS Securities Australia Ltd Buy 5,568 5,74 UBS Securities Australia Ltd Buy 5,568 5,74 UBS Securities Australia Ltd Buy 1,733 1,787 UBS Securities Australia Ltd Buy 1,523 1,704 UBS Securities Australia Ltd Buy 1,523 1,704 UBS Securities Australia Ltd Buy 1,522 1,704 UBS Securities Australia Ltd Buy 1,523 1,704 UBS Securities Australia Ltd Buy 1,520 1,504 UBS Securities Australia Ltd Buy 1,530 1,504 UBS Securities Australia Ltd Buy 1,334 1,64 UBS Securities Australia Ltd Buy 1,320 1,504 UBS Securities Australia Ltd Buy 1,325 1,104 UBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia Ltd	Buy	281	59	Ordinary
UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 5,257 542 UBS Securities Australia Ltd Buy 5,257 542 UBS Securities Australia Ltd Buy 1,766 1,213 UBS Securities Australia Ltd Buy 1,766 1,213 UBS Securities Australia Ltd Buy 1,6529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 1,532 1,704 UBS Securities Australia Ltd Buy 1,539 1,504 UBS Securities Australia Ltd Buy 3,739 3,85 UBS Securities Australia Ltd Buy 1,590 1,504 UBS Securities Australia Ltd Buy 3,739 3,85 UBS Securities Australia Ltd Buy 1,590 1,504 UBS Securities Australia Ltd Buy 5,735 1,64 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	89	7	Ordinary
UBS Securities Australia Ltd Buy 115,43 1190 UBS Securities Australia Ltd Buy 5,257 5,42 UBS Securities Australia Ltd Buy 5,568 5,74 UBS Securities Australia Ltd Buy 4,695 444 UBS Securities Australia Ltd Buy 1,733 1,787 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 1,599 1,505 UBS Securities Australia Ltd Buy 1,599 1,505 UBS Securities Australia Ltd Buy 1,333 1,416 UBS Securities Australia Ltd Buy 1,320 1,604 UBS Securities Australia Ltd <t< td=""><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>89</td><td>7</td><td>Ordinary</td></t<>	un-14	UBS Securities Australia Ltd	Buy	89	7	Ordinary
UBS Securities Australia Ltd Buy 5.257 5.42 UBS Securities Australia Ltd Buy 5,568 5,74 UBS Securities Australia Ltd Buy 1,766 1,213 UBS Securities Australia Ltd Buy 4,695 444 UBS Securities Australia Ltd Buy 1,734 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 1,530 1,930 UBS Securities Australia Ltd Buy 3,739 385 UBS Securities Australia Ltd Buy 1,930 1,064 UBS Securities Australia Ltd Buy 1,330 1,064 UBS Securities Australia Ltd Buy 6,227 6,22 UBS Securities Australia Ltd Buy 6,227 6,22 UBS Securities Australia Ltd Buy 6,227 1,664 UBS Securities Australia Ltd Bu	un-14	UBS Securities Australia Ltd	Buy	11,543	1,190	Ordinary
UBS Securities Australia Ltd Buy 5,68 574 UBS Securities Australia Ltd Buy 5,68 5,74 UBS Securities Australia Ltd Buy 11,766 1,213 UBS Securities Australia Ltd Buy 16,229 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 1,530 1,504 UBS Securities Australia Ltd Buy 3,735 385 UBS Securities Australia Ltd Buy 1,530 1,64 UBS Securities Australia Ltd Buy 3,735 3,73 UBS Securities Australia Ltd Buy 1,330 1,64 UBS Securities Australia Ltd Buy 6,208 6,208 UBS Securities Australia Ltd Buy 6,200 6,208 UBS Securities Australia Ltd Buy 6,200 1,104 UBS Securities Australia Ltd Bu	un-14	UBS Securities Australia Ltd	Buy	5,257	542	Ordinary
UBS Securities Australia Ltd Buy 5,568 574 UBS Securities Australia Ltd Buy 1,766 1,213 UBS Securities Australia Ltd Buy 1,334 1,767 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 1,599 1,505 UBS Securities Australia Ltd Buy 3,735 1,865 UBS Securities Australia Ltd Buy 1,599 1,505 UBS Securities Australia Ltd Buy 1,539 1,604 UBS Securities Australia Ltd Buy 1,335 1,416 UBS Securities Australia Ltd Buy 1,335 1,416 UBS Securities Australia Ltd Buy 1,330 1,505 UBS Securities Australia Ltd Buy 5,008 6,008 UBS Securities Australia Ltd Buy 5,008 6,008 UBS Securities Australia Ltd Buy 7,139 1,612 UBS Securities Australia Ltd <	un-14	UBS Securities Australia Ltd	Buy	78	8	Ordinary
UBS Securities Australia Ltd Buy 1,756 1,213 UBS Securities Australia Ltd Buy 4,695 484 UBS Securities Australia Ltd Buy 1,734 1,787 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 1,529 1,505 UBS Securities Australia Ltd Buy 3,735 385 UBS Securities Australia Ltd Buy 1,330 1,99 UBS Securities Australia Ltd Buy 1,330 1,99 UBS Securities Australia Ltd Buy 1,320 1,064 UBS Securities Australia Ltd Buy 1,320 1,064 UBS Securities Australia Ltd Buy 6,227 6,42 UBS Securities Australia Ltd Buy 6,227 2,083 UBS Securities Australia Ltd Buy 6,227 2,083 UBS Securities Australia Ltd Buy 7,139 7,61 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	5,568	574	Ordinary
UBS Securities Australia Ltd Buy 14695 484 UBS Securities Australia Ltd Buy 17,334 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 5,384 5,55 UBS Securities Australia Ltd Buy 5,384 5,55 UBS Securities Australia Ltd Buy 1,590 1,500 UBS Securities Australia Ltd Buy 1,390 1,99 UBS Securities Australia Ltd Buy 1,390 1,99 UBS Securities Australia Ltd Buy 3,73 3,73 UBS Securities Australia Ltd Buy 6,20 1,00 UBS Securities Australia Ltd Buy 6,20 1,00 UBS Securities Australia Ltd Buy 6,20 2,00 UBS Securities Australia Ltd Buy 6,20 2,00 UBS Securities Australia Ltd Buy 6,20 2,119 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	11,766	1,213	Ordinary
UBS Securities Australie Ltd Buy 17.334 1.787 UBS Securities Australie Ltd Buy 16.529 1.704 UBS Securities Australia Ltd Buy 16.529 1.704 UBS Securities Australia Ltd Buy 5.384 5.58 UBS Securities Australia Ltd Buy 3.279 38 UBS Securities Australia Ltd Buy 1,505 1,505 UBS Securities Australia Ltd Buy 1,532 1,416 UBS Securities Australia Ltd Buy 1,532 1,416 UBS Securities Australia Ltd Buy 1,622 62 UBS Securities Australia Ltd Buy 340 15 UBS Securities Australia Ltd Buy 6,227 6,208 UBS Securities Australia Ltd Buy 6,7008 6,908 UBS Securities Australia Ltd Buy 7,139 1,61 UBS Securities Australia Ltd Buy 6,7008 6,908 UBS Securities Australia Ltd Buy 7,139 1,612 UBS Securities Australia Ltd Buy </td <td>nn-14</td> <td>UBS Securities Australia Ltd</td> <td>Виу</td> <td>4,695</td> <td>484</td> <td>Ordinary</td>	nn-14	UBS Securities Australia Ltd	Виу	4,695	484	Ordinary
UBS Securities Australia Ltd Buy 15.529 1,704 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 1,529 1,704 UBS Securities Australia Ltd Buy 3,739 3,88 UBS Securities Australia Ltd Buy 1,599 1,505 UBS Securities Australia Ltd Buy 1,330 199 UBS Securities Australia Ltd Buy 1,373 146 UBS Securities Australia Ltd Buy 1,031 1,064 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy 7,000 700 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia Ltd	Buy	17,334	1,787	Ordinary
UBS Securities Australia Ltd Buy 18.9 UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 5,384 5,55 UBS Securities Australia Ltd Buy 1,459 1,505 UBS Securities Australia Ltd Buy 1,459 1,505 UBS Securities Australia Ltd Buy 1,735 1,416 UBS Securities Australia Ltd Buy 1,737 1,064 UBS Securities Australia Ltd Buy 1,0321 1,064 UBS Securities Australia Ltd Buy 6,227 6,22 UBS Securities Australia Ltd Buy 6,208 1,064 UBS Securities Australia Ltd Buy 6,208 1,164 UBS Securities Australia Ltd Buy 6,208 2,020 UBS Securities Australia Ltd Buy 7,139 7,36 UBS Securities Australia Ltd Buy 7,139 7,36 UBS Securities Australia Ltd Buy 7,139 7,135 UBS Securities Australia Ltd Buy 7,	un-14	UBS Securities Australia Ltd	Buy	16,529	1,704	Ordinary
UBS Securities Australia Ltd Buy 16,529 1,704 UBS Securities Australia Ltd Buy 5,884 555 UBS Securities Australia Ltd Buy 3,279 388 UBS Securities Australia Ltd Buy 1,599 1,505 UBS Securities Australia Ltd Buy 1,373 1,416 UBS Securities Australia Ltd Buy 1,527 1,064 UBS Securities Australia Ltd Buy 340 1,56 UBS Securities Australia Ltd Buy 3,70 1,064 UBS Securities Australia Ltd Buy 6,227 6,2 UBS Securities Australia Ltd Buy 6,008 6,908 UBS Securities Australia Ltd Buy 7,139 1,6 UBS Securities Australia Ltd Buy 1,6 2,185 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	18,023	1,858	Ordinary
UBS Securities Australia Ltd Buy 5.384 555 UBS Securities Australia Ltd Buy 3,735 338 UBS Securities Australia Ltd Buy 1,505 1,505 UBS Securities Australia Ltd Buy 1,733 1,93 UBS Securities Australia Ltd Buy 1,330 100 UBS Securities Australia Ltd Buy 1,031 100 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy 7,139 76 UBS Securities Australia Ltd Buy 1,612 2,165 UBS Securities Australia Ltd Buy 1,612 2,165 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	16,529	1,704	Ordinary
UBS Securities Australia Ltd Buy 3.279 338 UBS Securities Australia Ltd Buy 14,599 1,505 UBS Securities Australia Ltd Buy 3,735 385 1,605 UBS Securities Australia Ltd Buy 1,320 1,930 1,032 1,042 <th< td=""><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>5,384</td><td>555</td><td>Ordinary</td></th<>	un-14	UBS Securities Australia Ltd	Buy	5,384	555	Ordinary
UBS Securities Australia Ltd Buy 14,599 1,505 UBS Securities Australia Ltd Buy 1,439 1,905 UBS Securities Australia Ltd Buy 1,932 1,416 UBS Securities Australia Ltd Buy 1,0321 1,064 UBS Securities Australia Ltd Buy 9,70 1,004 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,7068 6,908 UBS Securities Australia Ltd Buy 6,7068 6,908 UBS Securities Australia Ltd Buy 7,139 1,315 UBS Securities Australia Ltd Buy 7,139 1,612 UBS Securities Australia Ltd Buy 7,139 1,612 UBS Securities Australia Ltd Buy 1,512 2,185 UBS Securities Australia Ltd Buy 1,612 2,90 UBS Securities Australia Ltd Buy 1,612 2,90 UBS Securities Australia Ltd Buy 4,249 4,88 UBS Securities Australia Ltd <td< td=""><td>un-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,279</td><td>338</td><td>Ordinary</td></td<>	un-14	UBS Securities Australia Ltd	Buy	3,279	338	Ordinary
UBS Securities Australia Ltd Buy 3,735 385 UBS Securities Australia Ltd Buy 1,930 199 UBS Securities Australia Ltd Buy 1,0221 1,064 UBS Securities Australia Ltd Buy 340 100 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy 7,139 756 UBS Securities Australia Ltd Buy 7,139 76 UBS Securities Australia Ltd Buy 1,195 2,185 UBS Securities Australia Ltd Buy 1,512 2,185 UBS Securities Australia Ltd Buy 1,512 2,185 UBS Securities Australia Ltd Buy 1,523 1,612 UBS Securities Australia Ltd Buy 4,249 4,28 UBS Securities Australia Ltd Buy <	un-14	UBS Securities Australia Ltd	Buy	14,599	1,505	Ordinary
UBS Securities Australia Ltd Buy 1,930 199 UBS Securities Australia Ltd Buy 13,735 1,416 UBS Securities Australia Ltd Buy 970 1,00 UBS Securities Australia Ltd Buy 340 35 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,708 1,0 UBS Securities Australia Ltd Buy 7,139 7,36 UBS Securities Australia Ltd Buy 2,1,195 7,16 UBS Securities Australia Ltd Buy 6,955 7,17 UBS Securities Australia Ltd Buy 6,955 7,17 UBS Securities Australia Ltd Buy 4,249 4,249 UBS Securities Australia Ltd Buy 4,249 4,28 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	3,735	382	Ordinary
UBS Securities Australia Ltd Buy 13,735 1,416 UBS Securities Australia Ltd Buy 10,321 1,064 UBS Securities Australia Ltd Buy 970 100 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,208 6,908 UBS Securities Australia Ltd Buy 6,7008 6,908 UBS Securities Australia Ltd Buy 7,139 136 UBS Securities Australia Ltd Buy 7,139 136 UBS Securities Australia Ltd Buy 7,139 1,612 UBS Securities Australia Ltd Buy 7,139 1,612 UBS Securities Australia Ltd Buy 6,955 7,17 UBS Securities Australia Ltd Buy 1,623 1,612 UBS Securities Australia Ltd Buy 1,636 1,612 UBS Securities Australia Ltd Buy 1,232 1,612 UBS Securities Australia Ltd Buy 4,249 4,28 UBS Securities Australia Ltd Buy <td>un-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>1,930</td> <td>199</td> <td>Ordinary</td>	un-14	UBS Securities Australia Ltd	Buy	1,930	199	Ordinary
UBS Securities Australia Ltd Buy 10,321 1,064 UBS Securities Australia Ltd Buy 340 375 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,207 6,42 UBS Securities Australia Ltd Buy 7,139 7,36 UBS Securities Australia Ltd Buy 7,139 1,36 UBS Securities Australia Ltd Buy 7,139 1,185 UBS Securities Australia Ltd Buy 6,955 7,18 UBS Securities Australia Ltd Buy 6,955 7,18 UBS Securities Australia Ltd Buy 1,612 20 UBS Securities Australia Ltd Buy 1,623 1,612 UBS Securities Australia Ltd Buy 1,532 1,612 UBS Securities Australia Ltd Buy 4,249 4,28 UBS Securities Australia Ltd Buy 4,249 4,28 UBS Securities Australia Ltd Buy	un-14	UBS Securities Australia Ltd	Buy	13.735	1.416	Ordinary
UBS Securities Australia Ltd Buy 970 100 UBS Securities Australia Ltd Buy 970 100 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,708 6,908 UBS Securities Australia Ltd Buy 7,139 7,88 UBS Securities Australia Ltd Buy 7,139 7,88 UBS Securities Australia Ltd Buy 7,139 7,185 UBS Securities Australia Ltd Buy 7,195 2,185 UBS Securities Australia Ltd Buy 6,955 7,77 UBS Securities Australia Ltd Buy 6,955 7,77 UBS Securities Australia Ltd Buy 6,955 7,77 UBS Securities Australia Ltd Buy 1,612 2,9 UBS Securities Australia Ltd Buy 4,249 4,38 UBS Securities Australia Ltd Buy 4,249 4,38 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy <	un-14	UBS Securities Australia Ltd	Buy	10.321	1.064	Ordinary
UBS Securities Australia Ltd Buy 340 35 UBS Securities Australia Ltd Buy 6.227 642 UBS Securities Australia Ltd Buy 146 15 UBS Securities Australia Ltd Buy 67,008 6,908 UBS Securities Australia Ltd Buy 7,139 736 UBS Securities Australia Ltd Buy 7,139 736 UBS Securities Australia Ltd Buy 21,195 2,185 UBS Securities Australia Ltd Buy 6,955 71 UBS Securities Australia Ltd Buy 1,612 29 UBS Securities Australia Ltd Buy 1,623 1,612 UBS Securities Australia Ltd Buy 1,232 1,612 UBS Securities Australia Ltd Buy 4,243 488 UBS Securities Australia Ltd Buy 4,249 488 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 264 UBS Securities Australia Ltd Buy 2,56	41-00	IUBS Securities Australia Ltd	Buv	026	100	Ordinary
UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 6,227 642 UBS Securities Australia Ltd Buy 67,008 6,908 UBS Securities Australia Ltd Buy 20,055 2,083 UBS Securities Australia Ltd Buy 7,139 7,36 UBS Securities Australia Ltd Buy 2,1,195 2,185 UBS Securities Australia Ltd Buy 2,1,195 2,185 UBS Securities Australia Ltd Buy 6,955 7,17 UBS Securities Australia Ltd Buy 1,5,636 1,612 UBS Securities Australia Ltd Buy 1,232 1,612 UBS Securities Australia Ltd Buy 4,249 4,28 UBS Securities Australia Ltd Buy 4,249 4,28 UBS Securities Australia Ltd Buy 4,249 4,28 UBS Securities Australia Ltd Buy 2,1990 2,567 UBS Securities Australia Ltd Buy 2,1990 3,67 UBS Securities Australia Ltd <	114	LIBS Securities Australia I td	B. B.	340	3 K	Ordinan
UBS Securities Australia Ltd Buy 67,22 UBS Securities Australia Ltd Buy 67,008 6,908 UBS Securities Australia Ltd Buy 20,205 2,083 UBS Securities Australia Ltd Buy 7,1319 736 UBS Securities Australia Ltd Buy 21,195 2,185 UBS Securities Australia Ltd Buy 6,965 77 UBS Securities Australia Ltd Buy 6,965 77 UBS Securities Australia Ltd Buy 6,965 77 UBS Securities Australia Ltd Buy 1,612 29 UBS Securities Australia Ltd Buy 1,623 1,612 UBS Securities Australia Ltd Buy 1,232 1,77 UBS Securities Australia Ltd Buy 4,249 4,38 UBS Securities Australia Ltd Buy 4,249 4,38 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 264 UBS Securities Australia Ltd Buy 2,567	14-01	I IRS Securities Australia 14d	8187	250	509	Ordinary Ordinary
UBS Securities Australia Ltd Buy 67,008 6,908 UBS Securities Australia Ltd Buy 20,205 2,083 UBS Securities Australia Ltd Buy 7,139 736 UBS Securities Australia Ltd Buy 7,195 2,185 UBS Securities Australia Ltd Buy 21,195 2,185 UBS Securities Australia Ltd Buy 6,955 77 UBS Securities Australia Ltd Buy 1,612 29 UBS Securities Australia Ltd Buy 1,232 1,612 UBS Securities Australia Ltd Buy 4,249 488 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy	110-14	LIRS Securities Australia Ltd	Birk	146	1,00	Ordinan
UBS Securities Australie Ltd Buy 20,005 2,083 UBS Securities Australie Ltd Buy 7,139 736 UBS Securities Australia Ltd Buy 7,139 136 UBS Securities Australia Ltd Buy 21,195 2,185 UBS Securities Australia Ltd Buy 6,955 7,17 UBS Securities Australia Ltd Buy 1,5,636 1,612 UBS Securities Australia Ltd Buy 1,5,336 1,612 UBS Securities Australia Ltd Buy 1,232 127 UBS Securities Australia Ltd Buy 4,249 4,88 UBS Securities Australia Ltd Buy 2,1990 2,267 UBS Securities Australia Ltd Buy 4,249 4,88 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,561 2,64	11-14	UBS Securities Australia IId	Brv	870 29	800 9	Ordinan
UBS Securities Australia Ltd Buy 7,139 736 UBS Securities Australia Ltd Buy 7,139 736 UBS Securities Australia Ltd Buy 2,195 2,185 UBS Securities Australia Ltd Buy 6,955 717 UBS Securities Australia Ltd Buy 6,955 717 UBS Securities Australia Ltd Buy 1,612 281 UBS Securities Australia Ltd Buy 1,232 1,712 UBS Securities Australia Ltd Buy 4,249 438 UBS Securities Australia Ltd Buy 4,249 438 UBS Securities Australia Ltd Buy 2,267 367 UBS Securities Australia Ltd Buy 2,267 367 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 2,567 UBS Securities Australia Ltd Buy 2,567 2,567	14	I IBS Securities Australia 14d	N. B.	30.005	580 6	Ordinan
UBS Securities Australia Ltd Buy 1,319 1,55 UBS Securities Australia Ltd Buy 1,319 1,185 UBS Securities Australia Ltd Buy 6,955 7,7 UBS Securities Australia Ltd Buy 6,955 7,7 UBS Securities Australia Ltd Buy 6,955 7,7 UBS Securities Australia Ltd Buy 1,612 29 UBS Securities Australia Ltd Buy 1,232 1,27 UBS Securities Australia Ltd Buy 4,249 488 UBS Securities Australia Ltd Buy 4,249 488 UBS Securities Australia Ltd Buy 4,249 488 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 367	14	LIBS Securities Australia 14d	6.8	7 139	257	Ordinary
UBS Securities Australia Ltd Buy 21,195 2,185 UBS Securities Australia Ltd Buy 6,955 7 UBS Securities Australia Ltd Buy 6,955 7/1 UBS Securities Australia Ltd Buy 15,636 1,612 UBS Securities Australia Ltd Buy 1,232 29 UBS Securities Australia Ltd Buy 4,249 488 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,1990 2,267 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,561 2,67 UBS Securities Australia Ltd Buy 2,561 2,67	14	I IBS Securities Australia I to	Bit	1310	35,	Ordinary
UBS Securities Australia Ltd Buy 68 7 UBS Securities Australia Ltd Buy 6.955 717 UBS Securities Australia Ltd Buy 15.636 1.612 UBS Securities Australia Ltd Buy 281 1.23 UBS Securities Australia Ltd Buy 4,249 4.38 UBS Securities Australia Ltd Buy 2,1990 2,267 UBS Securities Australia Ltd Buy 3,560 367 UBS Securities Australia Ltd Buy 2,561 244 UBS Securities Australia Ltd Buy 2,561 2,561	41-nu	UBS Securities Australia Ltd	Biv	21 195	2.185	Ordinary
UBS Securities Australia Ltd Buy 6,955 717 UBS Securities Australia Ltd Buy 1,612 1,612 UBS Securities Australia Ltd Buy 1,232 127 UBS Securities Australia Ltd Buy 4,249 438 UBS Securities Australia Ltd Buy 4,249 438 UBS Securities Australia Ltd Buy 2,1990 2,267 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,561 244	un-14	UBS Securities Australia Ltd	Birv	89	7	Ordinary
UBS Securities Australia Ltd Buy 15,036 1,612 UBS Securities Australia Ltd Buy 281 29 UBS Securities Australia Ltd Buy 1,232 127 UBS Securities Australia Ltd Buy 4,249 438 UBS Securities Australia Ltd Buy 2,1990 2,267 UBS Securities Australia Ltd Buy 3,560 367 UBS Securities Australia Ltd Buy 2,561 264	un-14	UBS Securities Australia Ltd	S S	6 955	717	Ordinary
UBS Securities Australia Ltd Buy 281 29 UBS Securities Australia Ltd Buy 1,232 127 UBS Securities Australia Ltd Buy 4,249 438 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,561 264 UBS Securities Australia Ltd Buy 2,561 264	un-14	UBS Securities Australia Ltd	Si.	15,636	1,612	Ordinary
UBS Securities Australia Ltd Buy 1,232 127 UBS Securities Australia Ltd Buy 4,249 438 UBS Securities Australia Ltd Buy 2,1990 2,267 UBS Securities Australia Ltd Buy 2,567 367 UBS Securities Australia Ltd Buy 2,561 244	1n-14	LIBS Securities Australia Itd	Bes	281	02	Ordinany
UBS Securities Australia td Buy 21,590 2,677 UBS Securities Australia td Buy 21,590 2,677 UBS Securities Australia td Buy 2,567 UBS Securities Australia td Buy 2,561 2,671 UBS Securities Australia td Buy 2,561 2,671	14	LIBS Societies Australia 14d	600	1 232	437	Ordinary
UBS Securities Australia Ltd Buy 21,990 2,677 UBS Securities Australia Ltd Buy 21,990 2,677 UBS Securities Australia Ltd Buy 3,560 367 UBS Securities Australia Ltd Buy 2,561 264	1 2	LIBS Securities Australia ted	Sod o	262,1	171	Ordinary
UPS Securities Australia Ltd Buy 1,1950 2,267 UBS Securities Australia Ltd Buy 3,560 367 UBS Securities Australia Ltd Buy 2,561 264	4	UDS Securities Australia Ltd	Buy	4,249	438	Ordinary
UBS Securities Australia Ltd Buy 2,560 367 264	41:14	UBS Securities Australia Ltd	Buy	21,990	2,267	Ordinary
UBS Securities Australia Ltd Buy 2,561 264	4-14	UBS Securities Australia Ltd	Buy	3.560	192	Ordinary
	71-4					Cramany

aria aria
BG 2
Buy
Buy
<u> </u>
3
3
Sell
1 8
B
<u> </u>
<u> </u>
2 3
3
3
3
3 3
2
<u> </u>
ž (
9,
g g
es.
Sell
= Se
Sell
Sei
<u>-</u>
ق
3
5
ķ .
<u> </u>
1
Ŕ
Š
<u>R</u>
<u>B</u>
8
Sel
Se S
Sell
Š
Ş
و
3 3
ķ .
<u></u>
<u></u>
Sell
Se
Sel
اوا
3
ž
es S
<u>8</u>
3
<u> </u>
S
9
3
ቜ
Je Se
Sell
3
ķ
3

URS Securities Australia Ltd Buy 25,147 URS Securities Australia Ltd Buy 25,007 URS Securities Australia Ltd Buy 2,007 URS Securities Australia Ltd Buy 1,661 URS Securities Australia Ltd Buy 1,661 URS Securities Australia Ltd Buy 49,147 URS Securities Australia Ltd Buy 49,150 URS Securities Australia Ltd Buy 1,920 URS Securities Australia Ltd Buy 1,920 URS Securities Australia Ltd Buy 1,920 URS Securities Australia Ltd Buy 1,166 URS Securities Australia Ltd Buy	2 hm 14	I IDC Committee Assessing to a				2000
USS Scornites Australia Ltd BM 2,001 USS Scornites Australia Ltd BM 4,04 USS Scornites Australia Ltd BM 4,148 USS Scornites Australia Ltd BM 4,148 USS Scornites Australia Ltd BM 1,150 USS Scornites Australia Ltd BM 7,153 USS Scornites Australia Ltd BM 7,150 USS Scornites Australia Ltd BM 1,150 USS Scornites Australia Ltd BM 3,146	110-14	UBS Securities Australia Ind	Bary	25 302	2,095	Ordinary
UBS Securities Australia Ltd Buy 14 8.92 UBS Securities Australia Ltd Buy 4.1,306 UBS Securities Australia Ltd Buy 7.5.38 UBS Securities Australia Ltd Buy 7.7.538 UBS Securities Australia Ltd Buy 7.7.847 UBS Securities Australia Ltd Buy 7.7.847 UBS Securities Australia Ltd Buy 7.7.847 UBS Securities Australia Ltd Buy 3.1.46 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Bity	2,007	2007	Ordinary
USS Securities Australia Ltd Buy 44,147 USS Securities Australia Ltd Buy 14,206 USS Securities Australia Ltd Buy 7,143 USS Securities Australia Ltd Buy 7,143 USS Securities Australia Ltd Buy 7,150 USS Securities Australia Ltd Buy 1,200 USS Securities Australia Ltd Buy 1,150 USS Securities Australia Ltd Buy 3,146 USS Securities Australia Ltd Buy	Jun-14	I IRS Securities Australia Ltd	B	14.842	1 530	Ordina,
UBS Securities Australia Ltd Buy 457,238 UBS Securities Australia Ltd Buy 457,238 UBS Securities Australia Ltd Buy 51,438 UBS Securities Australia Ltd Buy 51,432 UBS Securities Australia Ltd Buy 77,538 UBS Securities Australia Ltd Buy 7,530 UBS Securities Australia Ltd Buy 7,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy	1 1	Libe Consider Australia Ltu	ing.	14,642	000,1	Ordinary
URS Securities Australia Ltd Buy 18,14 URS Securities Australia Ltd Buy 18,14 URS Securities Australia Ltd Buy 77,538 URS Securities Australia Ltd Buy 77,538 URS Securities Australia Ltd Buy 1,190 URS Securities Australia Ltd Buy 3,146 URS Securities Australia Ltd Buy	1 1 1 1	CDD Securities Australia Ltd	, and	04,40	3,555	Ordinary
UBS Securities Australia Ltd Buy 51.48 UBS Securities Australia Ltd Buy 51.48 UBS Securities Australia Ltd Buy 77.538 UBS Securities Australia Ltd Buy 77.538 UBS Securities Australia Ltd Buy 31.66 UBS Securities Australia Ltd Buy		Obs securities Australia Liu	Buy	49,147	5,093	Orginary
UBS Securities Australia Ltd Buy 77,538 UBS Securities Australia Ltd Buy 77,548 UBS Securities Australia Ltd Buy 77,548 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy	1	Ubb Securities Australia Ltd	Buy	(18,8)	1,929	Ordinary
UBS Securities Australia Ltd Buy 77.528 UBS Securities Australia Ltd Buy 17.528 UBS Securities Australia Ltd Buy 20.912 UBS Securities Australia Ltd Buy 3.18 UBS Securities Australia Ltd Buy 3.16 UBS Securities Australia Ltd Buy 3.16 </td <td>7UU-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>5,143</td> <td>533</td> <td>Ordinary</td>	7UU-14	UBS Securities Australia Ltd	Buy	5,143	533	Ordinary
UBS Securities Australia Ltd Buy 10.084 UBS Securities Australia Ltd Buy 1.920 UBS Securities Australia Ltd Buy 3.146 UBS Securities Australia Ltd Buy	100-14 14	UBS Securities Australia Ltd	Buy	17,538	8,035	Ordinary
UBS Securities Australia Ltd Buy 1,020 IL UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 2,720 UBS Securities Australia Ltd Buy 2,720 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	10,084	1,045	Ordinary
UBS Securities Australia Ltd Buy 1,920 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	20,912	2,167	Ordinary
UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 1,7467 UBS Securities Australia Ltd Buy 1,7467 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	1,920	199	Ordinary
UBS Securities Australia Ltd Buy 318 UBS Securities Australia Ltd Buy 3146 UBS Securities Australia Ltd Buy 3146 UBS Securities Australia Ltd Buy 7.7 840 UBS Securities Australia Ltd Buy 2.7 840 UBS Securities Australia Ltd Buy 3.146 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	4,150	430	Ordinary
UBS Securities Australia Ltd Buy 3146 UBS Securities Australia Ltd Buy 3146 UBS Securities Australia Ltd Buy 7.847 UBS Securities Australia Ltd Buy 7.7867 UBS Securities Australia Ltd Buy 3.146 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	318	33	Ordinary
UBS Securities Australia Ltd Buy 178 UBS Securities Australia Ltd Buy 178 UBS Securities Australia Ltd Buy 27.840 UBS Securities Australia Ltd Buy 27.840 UBS Securities Australia Ltd Buy 3.146 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Biv	3.146	325	Ordinary
UBS Securities Australia Ltd Buy 17,467 UBS Securities Australia Ltd Buy 17,467 UBS Securities Australia Ltd Buy 17,467 UBS Securities Australia Ltd Buy 31,46 UBS Securities Australia Ltd Buy	1 n-14	I IRS Securities Australia Ltd	Bis	218	27	Ording
UBS Securities Australia Ltd Buy 17,400 UBS Securities Australia Ltd Buy 27,840 UBS Securities Australia Ltd Buy 27,840 UBS Securities Australia Ltd Buy 31.46 UBS Securities Australia Ltd Buy	10.1	LIPS Countries Australia Library	600	237.51	1 010	i i
UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy		ODS Securities Australia Ltd	Ang	1,40/	010,	Ordinary
UBS Securities Australia Ltd Buy 4150 UBS Securities Australia Ltd Buy 31.46 UBS Securities Australia Ltd Buy	100	Ubb Securities Australia Ltd	Buy	27,840	7,885	Ordinary
UBS Securities Australia Ltd Buy 31.46 UBS Securities Australia Ltd Buy 3.146 UBS Securities Australia Ltd Buy 5.02 UBS Securities Australia Ltd Buy 5.2349 UBS Securities Australia Ltd Buy	tr-in	UBS Securities Australia Ltd	Buy	4,150	430	Ordinary
UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 3 146 UBS Securities Australia Ltd Buy 2 234 UBS Securities Australia Ltd Buy 2 234 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	174	18	Ordinary
UBS Securities Australia Ltd Buy 3.831 UBS Securities Australia Ltd Buy 3.146 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	3,146	326	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 2,23 UBS Securities Australia Ltd Buy 2,13 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	3,831	397	Ordinary
UBS Securities Australia Ltd Buy 3.146 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 2,633 UBS Securities Australia Ltd Buy 2,630 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,234 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	3,146	326	Ordinary
UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 4,650 UBS Securities Australia Ltd Buy 5,636 UBS Securities Australia Ltd Buy 5,2349 UBS Securities Australia Ltd Buy 5,220 UBS Securities Australia Ltd Buy 2,136 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	3,146	326	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,590 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	4,150	430	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 5,620 UBS Securities Australia Ltd Buy 5,620 UBS Securities Australia Ltd Buy 5,620 UBS Securities Australia Ltd Buy 2,385 UBS Securities Australia Ltd Buy 2,385 UBS Securities Australia Ltd Buy 2,394 UBS Securities Australia Ltd Buy 3,165 UBS Securities Australia Ltd Buy 3,165 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buv	3,146	326	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,630 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 7,73 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buv	3.146	326	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 5,433 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 2,835 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 3,766 UBS Securities Australia Ltd Buy 3,760 UBS Securities Australia Ltd Buy	lin-14	I IRS Securities Australia Ltd	Bliv	3.176	328	Ordinan
UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 5,636 UBS Securities Australia Ltd Buy 5,733 UBS Securities Australia Ltd Buy 5,734 UBS Securities Australia Ltd Buy 6,734 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,139 UBS Securities Australia Ltd Buy	14	1 100 Consistent Australia Lea		2110	220	y in in
UBS Securities Australia Ltd Buy 4,150 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 7,431 UBS Securities Australia Ltd Buy 5,625 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy		Obs securities Australia Ltd	pony	3,140	370	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 7,71 UBS Securities Australia Ltd Buy 5,625 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 5,627 UBS Securities Australia Ltd Buy 5,629 UBS Securities Australia Ltd Buy 2,334 UBS Securities Australia Ltd Buy 3,362 UBS Securities Australia Ltd Buy	- Id	UBS Securities Australia Ltd	Buy	4,150	430	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 5,636 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,133 UBS Securities Australia Ltd Buy 2,133 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	3,146	326	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 7,431 UBS Securities Australia Ltd Buy 7,626 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 2,895 UBS Securities Australia Ltd Buy 2,334 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	3,146	326	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 5,433 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 2,835 UBS Securities Australia Ltd Buy 2,349 UBS Securities Australia Ltd Buy 2,349 UBS Securities Australia Ltd Buy 7,371 UBS Securities Australia Ltd Buy 7,360 UBS Securities Australia Ltd Buy 7,372 UBS Securities Australia Ltd Buy 3,368 UBS Securities Australia Ltd Buy 3,368 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,120 UBS Securities Australia Ltd Buy 2,107 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	3,146	326	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 7,433 UBS Securities Australia Ltd Buy 6,637 UBS Securities Australia Ltd Buy 6,734 UBS Securities Australia Ltd Buy 2,835 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,673 UBS Securities Australia Ltd Buy 2,663 UBS Securities Australia Ltd Buy	Ju-14	UBS Securities Australia Ltd	Buv	3.146	326	Ordinary
UBS Securities Australia Ltd Buy 3,146 UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 7,431 UBS Securities Australia Ltd Buy 7,71 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 5,627 UBS Securities Australia Ltd Buy 5,629 UBS Securities Australia Ltd Buy 2,839 UBS Securities Australia Ltd Buy 2,334 UBS Securities Australia Ltd Buy 6,794 UBS Securities Australia Ltd Buy 7,7003 UBS Securities Australia Ltd Buy 3,368 UBS Securities Australia Ltd Buy 3,113 UBS Securities Australia Ltd Buy 3,1671 UBS Securities Australia Ltd Buy 2,173 UBS Securities Australia Ltd Buy 2,173 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,667 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia I to	Silv	3 146	326	Ordinary
UBS Securities Australia Ltd Buy 4,690 UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 7,433 UBS Securities Australia Ltd Buy 5,433 UBS Securities Australia Ltd Buy 77 UBS Securities Australia Ltd Buy 6,626 UBS Securities Australia Ltd Buy 6,237 UBS Securities Australia Ltd Buy 22,349 UBS Securities Australia Ltd Buy 22,349 UBS Securities Australia Ltd Buy 77,44 UBS Securities Australia Ltd Buy 7,74 UBS Securities Australia Ltd Buy 7,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,100 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,500 UBS Securities Australia Ltd Buy	110-14	I IRS Securities Australia I to	Block	3146	376	Ordinan
UBS Securities Australia Ltd Buy 7,421 UBS Securities Australia Ltd Buy 7,431 UBS Securities Australia Ltd Buy 7,431 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 5,625 UBS Securities Australia Ltd Buy 2,835 UBS Securities Australia Ltd Buy 2,2349 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 3,760 UBS Securities Australia Ltd Buy 3,780 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,130 UBS Securities Australia Ltd Buy 2,200 UBS Securities Australia Ltd Buy 2,500 UBS Securities Australia Ltd Buy	2	I IBC Constitution Australia I + d		4 500	207	
UBS Securities Australia Ltd Buy 5,421 UBS Securities Australia Ltd Buy 5,431 UBS Securities Australia Ltd Buy 5,636 UBS Securities Australia Ltd Buy 5,637 UBS Securities Australia Ltd Buy 2,835 UBS Securities Australia Ltd Buy 2,834 UBS Securities Australia Ltd Buy 2,349 UBS Securities Australia Ltd Buy 7,703 UBS Securities Australia Ltd Buy 7,703 UBS Securities Australia Ltd Buy 7,736 UBS Securities Australia Ltd Buy 3,786 UBS Securities Australia Ltd Buy 3,787 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,130 UBS Securities Australia Ltd Buy 2,130 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,671 UBS Securities Australia Ltd Buy 2,673 UBS Securities Australia Ltd Buy		ODS Securities Australia LIG	, and	020,1	480	Ordinary
UBS Securities Australia Ltd Buy 5,433 UBS Securities Australia Ltd Buy 77 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 697 UBS Securities Australia Ltd Buy 697 UBS Securities Australia Ltd Buy 22,349 UBS Securities Australia Ltd Buy 22,349 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 73,603 UBS Securities Australia Ltd Buy 7,13 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,130 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,663 UBS Securities Australia Ltd Buy 2,663 UBS Securities Australia Ltd Buy 2,663 UBS Securities Australia Ltd Buy <th< td=""><td>1</td><td>Ubb Securities Australia Ltd</td><td>Buy</td><td>/,42</td><td>69/</td><td>Ordinary</td></th<>	1	Ubb Securities Australia Ltd	Buy	/,42	69/	Ordinary
UBS Securities Australia Ltd Buy 170 UBS Securities Australia Ltd Buy 77 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 16,337 UBS Securities Australia Ltd Buy 2,895 UBS Securities Australia Ltd Buy 2,2349 UBS Securities Australia Ltd Buy 6,794 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 3,368 UBS Securities Australia Ltd Buy 3,113 UBS Securities Australia Ltd Buy 3,157 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,130 UBS Securities Australia Ltd Buy 2,6673 UBS Securities Australia Ltd Buy 2,6673 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	5,433	263	Ordinary
UBS Securities Australia Ltd Buy 77 UBS Securities Australia Ltd Buy 5,626 UBS Securities Australia Ltd Buy 16,337 UBS Securities Australia Ltd Buy 1,835 UBS Securities Australia Ltd Buy 22,349 UBS Securities Australia Ltd Buy 2,734 UBS Securities Australia Ltd Buy 77,033 UBS Securities Australia Ltd Buy 7,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,130 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,500 UBS Securities Australia Ltd Buy 2,500 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	1,901	197	Ordinary
UBS securities Australia Ltd Buy 5.6.56 UBS Securities Australia Ltd Buy 6.97 UBS Securities Australia Ltd Buy 2.895 UBS Securities Australia Ltd Buy 2.23.49 UBS Securities Australia Ltd Buy 2.73.49 UBS Securities Australia Ltd Buy 7.75.620 UBS Securities Australia Ltd Buy 7.173 UBS Securities Australia Ltd Buy 2.113 UBS Securities Australia Ltd Buy 2.113 UBS Securities Australia Ltd Buy 2.113 UBS Securities Australia Ltd Buy 2.100 UBS Securities Australia Ltd Buy 2.100 UBS Securities Australia Ltd Buy 2.200 UBS Securities Australia Ltd Buy 15.305 UBS Securities Australia Ltd Buy 2.6673 UBS Securities Australia Ltd Buy 2.6673 UBS Securities Australia Ltd Buy 2.6673 UBS Securities Australia Ltd Buy 2.673 UBS Securities Australia Ltd Buy<	Jun-14	UBS Securities Australia Ltd	Buy	77	00	Ordinary
UBS Securities Australia Ltd Buy 897 UBS Securities Australia Ltd Buy 16.337 UBS Securities Australia Ltd Buy 2.895 UBS Securities Australia Ltd Buy 5.02 UBS Securities Australia Ltd Buy 5.734 UBS Securities Australia Ltd Buy 7.7033 UBS Securities Australia Ltd Buy 7.3,620 UBS Securities Australia Ltd Buy 7.113 UBS Securities Australia Ltd Buy 2.113 UBS Securities Australia Ltd Buy 2.735 UBS Securities Australia Ltd Buy 2.220 UBS Securities Australia Ltd Buy 2.220 UBS Securities Australia Ltd Buy 2.207 UBS Securities Australia Ltd Buy 2.507 UBS Securities Australia Ltd Buy 2.6,673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	5,626	583	Ordinary
UBS Securities Australia Ltd Buy 16,337 UBS Securities Australia Ltd Buy 502 UBS Securities Australia Ltd Buy 22,349 UBS Securities Australia Ltd Buy 22,349 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 3,1671 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,133 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,230 UBS Securities Australia Ltd Buy 2,505 UBS Securities Australia Ltd Buy 2,6,673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 6,500 1 UBS Securities Australia Ltd	Jun-14	UBS Securities Australia Ltd	Buv	897	93	Ordinary
UBS Securities Australia Ltd Buy 2,895 UBS Securities Australia Ltd Buy 2,895 UBS Securities Australia Ltd Buy 22,349 UBS Securities Australia Ltd Buy 6,794 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 3,368 UBS Securities Australia Ltd Buy 3,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,130 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,6673 UBS Securities Australia Ltd Buy 2,6673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 2,6673 UBS Securities Australia Ltd Buy 6,072 UBS Securities Australia Ltd Buy 6,072 UBS Securities Australia Ltd Buy	lun-14	UBS Securities Australia 11d	Birv	16 337	1 693	Ordinary
UBS Securities Australia Ltd Buy 502 UBS Securities Australia Ltd Buy 502 UBS Securities Australia Ltd Buy 6,794 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,133 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,230 UBS Securities Australia Ltd Buy 2,230 UBS Securities Australia Ltd Buy 2,530 UBS Securities Australia Ltd Buy 2,6,673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 6,573 UBS Securities Australia Ltd Buy 6,673 UBS Securities Australia Ltd Buy	11	I IRC Congrition Australia 14d	650	3 80E	200	or in a
UBS Securities Australia Ltd Buy 2.0.2 UBS Securities Australia Ltd Buy 2.2.349 UBS Securities Australia Ltd Buy 6,794 UBS Securities Australia Ltd Buy 73.6.20 UBS Securities Australia Ltd Buy 3,368 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,200 UBS Securities Australia Ltd Buy 1,530 UBS Securities Australia Ltd Buy 1,5,305 UBS Securities Australia Ltd Buy 2,5,305 UBS Securities Australia Ltd Buy 2,6,73 UBS Securities Australia Ltd Buy 2,6,73 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 6,500 1 UBS Securities Australia Ltd Buy 6,500 1 UBS Securities Australia Ltd Buy 6,500 1		Did permitted Control	Social Control of the	569,2	200	C C C I say
UBS Securities Australia Ltd Buy 6.7349 UBS Securities Australia Ltd Buy 6.794 UBS Securities Australia Ltd Buy 7.003 UBS Securities Australia Ltd Buy 3.368 UBS Securities Australia Ltd Buy 2.113 UBS Securities Australia Ltd Buy 2.113 UBS Securities Australia Ltd Buy 2.220 UBS Securities Australia Ltd Buy 1.833 UBS Securities Australia Ltd Buy 2.220 UBS Securities Australia Ltd Buy 1.5305 UBS Securities Australia Ltd Buy 2.6673 UBS Securities Australia Ltd Buy 2.6673 UBS Securities Australia Ltd Buy 3.6673 UBS Securities Australia Ltd Buy 4.057 UBS Securities Australia Ltd Buy	# - I	USS Securities Australia Ltd	Buy	205	25	Orginary
UBS Securities Australia Ltd Buy 6,794 UBS Securities Australia Ltd Buy 17,003 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,133 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,230 UBS Securities Australia Ltd Buy 2,567 UBS Securities Australia Ltd Buy 2,6673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 5,6673 UBS Securities Australia Ltd Buy 5,6673 UBS Securities Australia Ltd Buy 5,6673	Jun-14	UBS Securities Australia Ltd	Buy	22,349	2,316	Ordinary
UBS Securities Australia Ltd Buy 17,003 UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 3,188 UBS Securities Australia Ltd Buy 3,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 1,833 UBS Securities Australia Ltd Buy 7,73 UBS Securities Australia Ltd Buy 2,5,05 UBS Securities Australia Ltd Buy 2,6,673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 5,6,573 UBS Securities Australia Ltd Buy 5,6,573 UBS Securities Australia Ltd Buy 5,6,673	Jun-14	UBS Securities Australia Ltd	Buy	6,794	704	Ordinary
UBS Securities Australia Ltd Buy 73,620 UBS Securities Australia Ltd Buy 3,888 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,133 UBS Securities Australia Ltd Buy 1,853 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,507 UBS Securities Australia Ltd Buy 2,6,673 UBS Securities Australia Ltd Buy 36,673 UBS Securities Australia Ltd Buy 4,075 UBS Securities Australia Ltd Buy 4,075 UBS Securities Australia Ltd Buy 4,070 UBS Securities Australia Ltd Buy 5,673	Jun-14	UBS Securities Australia Ltd	Buy	17,003	1,762	Ordinary
UBS Securities Australia Ltd Buy 3,368 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,200 UBS Securities Australia Ltd Buy 1,500 UBS Securities Australia Ltd Buy 1,1,291 UBS Securities Australia Ltd Buy 2,6,673 UBS Securities Australia Ltd Buy 2,6,673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 85,500 UBS Securities Australia Ltd Buy 5,673 UBS Securities Australia Ltd Buy 5,673	Jun-14	UBS Securities Australia Ltd	Buy	73,620	7,629	Ordinary
UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 31,671 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 1,853 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 15,305 UBS Securities Australia Ltd Buy 17,71 UBS Securities Australia Ltd Buy 26,673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 26,673 UBS Securities Australia Ltd Buy 26,673 UBS Securities Australia Ltd Buy 26,673	Jun-14	UBS Securities Australia Ltd	Buy	3.368	349	Ordinary
UBS Securities Australia Ltd Buy 31,671 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 1,853 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 2,230 UBS Securities Australia Ltd Buy 2,505 UBS Securities Australia Ltd Buy 26,673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 56,503	11m-14	I IBS Securities Australia ! td	200	2 113	210	Ordinary
UBS Securities Australia Ltd Buy 2,1071 UBS Securities Australia Ltd Buy 2,113 UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 1,5305 UBS Securities Australia Ltd Buy 1,231 UBS Securities Australia Ltd Buy 2,5673 UBS Securities Australia Ltd Buy 2,6673 UBS Securities Australia Ltd Buy 3,632 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 8,500 UBS Securities Australia Ltd Buy 2,6573 UBS Securities Australia Ltd Buy 2,6673	1	I DC Counties Australia Ltd	(ng	21 571	2000	Oldinaly
UBS Securities Australia Ltd Buy 1.13 UBS Securities Australia Ltd Buy 1.853 UBS Securities Australia Ltd Buy 2.220 UBS Securities Australia Ltd Buy 1.5,305 UBS Securities Australia Ltd Buy 2.6,673 UBS Securities Australia Ltd Buy 2.6,673 UBS Securities Australia Ltd Buy 36,342 UBS Securities Australia Ltd Buy 4,073 UBS Securities Australia Ltd Buy 5,5673 UBS Securities Australia Ltd Buy 6,500 UBS Securities Australia Ltd Buy 6,500 UBS Securities Australia Ltd Buy 6,500	Juri-14	Obs securities Australia Ltd	eny	31,6/1	2,482	Orginary
UBS Securities Australia Ltd Buy 1,883 UBS Securities Australia Ltd Buy 2,200 UBS Securities Australia Ltd Buy 15,305 UBS Securities Australia Ltd Buy 11,201 UBS Securities Australia Ltd Buy 26,673 UBS Securities Australia Ltd Buy 36,342 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 56,73 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 56,73	Jun-14	UBS Securities Australia Ltd	Buy	2,113	219	Ordinary
UBS Securities Australia Ltd Buy 2,220 UBS Securities Australia Ltd Buy 15,305 UBS Securities Australia Ltd Buy 77 UBS Securities Australia Ltd Buy 11,291 UBS Securities Australia Ltd Buy 26,73 UBS Securities Australia Ltd Buy 36,342 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 56,673 UBS Securities Australia Ltd Buy 56,673	Jun-14	UBS Securities Australia Ltd	Buy	1,853	192	Ordinary
UBS Securities Australia Ltd Buy 15,305 UBS Securities Australia Ltd Buy 77 UBS Securities Australia Ltd Buy 26,673 UBS Securities Australia Ltd Buy 36,342 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 2,6673 UBS Securities Australia Ltd Buy 2,6673	Jun-14	UBS Securities Australia Ltd	Buy	2,220	230	Ordinary
UBS Securities Australia Ltd Buy 77 UBS Securities Australia Ltd Buy 11,291 UBS Securities Australia Ltd Buy 26,673 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 56,673 UBS Securities Australia Ltd Buy 56,673	2-Jun-14	UBS Securities Australia Ltd	Buy	15,305	1,586	Ordinary
UBS Securities Australia Ltd Buy 11,291 UBS Securities Australia Ltd Buy 26,673 UBS Securities Australia Ltd Buy 36,342 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 56,673	Jun-14	UBS Securities Australia Ltd	Buy	177	8	Ordinary
UBS Securities Australia Ltd	lı m-14	It IRC Sportition Australia Itd	Ran	11 291	1 170	Ordinary
UBS Securities Australia Ltd Buy 20,013 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 96,500	,	tipo contino Australia Liu	ing o	162,11	27.7	Ordinary
UBS Securities Australia Ltd Buy 36,342 UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 56,673 UBS Securities Australia Ltd Buy 26,673	Jun-14	UBS Securities Australia Ltd	Buy	76,6/3	2,/64	Ordinary
UBS Securities Australia Ltd Buy 4,072 UBS Securities Australia Ltd Buy 95,500 UBS Securities Australia Ltd Buy 26,673 UPS Committee Australia Ltd Buy 26,673	Jun-14	UBS Securities Australia Ltd	Buy	36,342	3,766	Ordinary
UBS Securities Australia Ltd Buy 96,500 UBS Securities Australia Ltd Buy 56,673 Inne Securities Australia Ltd Buy 56,673	Jun-14	UBS Securities Australia Ltd	Buy	4,072	422	Ordinary
UBS Securities Australia Ltd Buy 26,673	Jun-14	UBS Securities Australia Ltd	Buy	96,500	10,000	Ordinary
	Jun-14	UBS Securities Australia Ltd	Buv	26.673	2.764	Ordinary
C C C C C C C C C C C C C C C C C C C		1 10 C + 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	65	5 (2/52		5
UBS Securities Australia Ltd Buy 4,072	Jun-14	CF - C C C C C C C C C C		. 573	ÇÇ,	
I IRS Sociatios Australia I to		ODD SCONINGS AUSTRAIN CIU	Buy	4,072	422	Ordinary

2 1 10 14	I IDS Contribute Australia Ltd	Coff	676	(30)	, in the second
2-Jun-14	UBS Securities Australia Ltd	Sel	1.673	(167)	Ordinary
un-14	UBS Securities Australia Ltd	Sell	7,776	(800)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Sell	89	6	Ordinary
2-Jun-14	UBS Securities Australia Ltd	lles	1.186	(122)	Ordinary
2-Jun-14	LIBS Securities Australia Ltd	leS.	27.778	(2,292)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Sell	17.924	(1,844)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Sell	11,207	(1,153)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Sell	1.545	(159)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Sell	7,348	(126)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Sell	3,237	(333)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Sell	659'2	(788)	Ordinary
2-jun-14	UBS Securities Australia Ltd	Sell	5,978	(615)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Sell	2,974	(306)	Ordinary
hn-14	UBS Securities Australia Ltd	Sell	4,568	(470)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Sell	16,660	(1,714)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Self	2,006	(515)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Seff	385	(101)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Seil	5,735	(230)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Setl	194	(20)	Ordinary
un-14	UBS Securities Australia Ltd	Sell	29,160	(3,000)	Ordinary
41-17	UBS Securities Australia Ltd	Sell	37,548	(3,863)	Ordinary
Z-Jun-14	UBS Securities Australia Ltd	Sell	4,180	(430)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Sell	5,/35	(963)	Ordinary
2-Jun-14	UBS Securities Australia Ltd		091,82	(3,000)	Ordinary
51-UN-71	UBS Securities Australia Ltd	- Nell	4,190	(430)	Ordinary
4	UBS Securities Australia Ltd	Sell	091,67	(3,000)	Ordinary
2-Jun-2	UBS Securities Australia Ltd	nec .	4,800	(000)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Sell	000	(16)	Ordinary
12-Jun-14	UBS Securities Australia Ltd) Sell	26,954	(5,7,3)	Ordinary
Jun-14	UBS Securities Australia Ltd	Sell	5,735	(290)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Sell	49	(2)	Ordinary
Jun-14	UBS Securities Australia Ltd	Sell	78	(8)	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Self	4,180	(430)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Sell	525	(54)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Sell	29,160	(3,000)	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Sell	1,098	(113)	Ordinary
un-14	UBS Securities Australia Ltd	Sell	34,078	(3,506)	Ordinary
un-14	UBS Securities Australia Ltd	Buy	77	80	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Виу	77	8	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Buy	77	8	Ordinary
un-14	UBS Securities Australia Ltd	Buy	77	8	Ordinary
2-jun-14	UBS Securities Australia Ltd	Buy	77	8	Ordinary
un-14	UBS Securities Australia Ltd	Buy	77	8	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	77	8	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	39	4	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	48	5	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Buy	77	88	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	1,467	152	Ordinary
2-hin-14	UBS Securities Australia Ltd	Buy	39.150	4.057	Ordinary
2-lun-14	UBS Securities Australia Ltd	Buv	20.516	2,126	Ordinary
2-4 m-14	I IRS Securities Australia 11d	Buy	6 862	1 022	Ordinacy
2-July 14	OCO Secondes Associated Life		11 763	1 210	Ordinary
1	ODS Securities Australia Etc.	600	507(11)	000	O'dirai y
5-Jun-7	UBS Securities Australia Ltd	ing	/10/7	607	Ordinary
4	UBS Securities Australia Ltd	Buy	587'1	133	Ordinary
2-Jun-14	UBS Securities Australia Ltd	any	10,480	3,086	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	2,142	222	Ordinary
un-14	UBS Securities Australia Ltd	Buy	2,326	241	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	2,258	234	Ordinary
un-14	UBS Securities Australia Ltd	Buy	16,888	1,750	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	2,626	583	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	1,515	157	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	4,922	510	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Buv	7 633	100	
			000	5	Ordinary
11 m.1	I 100 Countition Australia #10	Alla	15 208	1576	Ordinary

2-Jun-14	Coo Securities Australia Lito	Ang.	011,4	<u>.</u>	Ordinary
	I IRS Securities Australia Ltd	Bits	23 307	NCN C	Ordinan
-Jun-14	UBS Securities Australia Ltd	Buv	4.265	447	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	174	1 8	Ordinary
7-lim-14	UBS Securities Australia Hd	Bus	900	90	ording.
2-Jun-14	UBS Securities Australia Ltd	Silv	174	18	Ordina
2-Jun-14	UBS Securities Australia Ltd	Buv	174	18	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buv	984	103	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	685	71	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	23,112	2,395	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buv	25.138	2,605	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	20.419	2.116	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	2,963	307	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	405	42	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Buy	21.278	2.205	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buv	1,255	130	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buv	11.291	1.170	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buv	83.965	8 701	Ordinary
-Jun-14	UBS Securities Australia Ltd	Suv	5 790	900	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	19.300	2.000	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buv	786	102	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	33	4	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Riv	174	αt	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	174	18	Ordinary
-Jun-14	UBS Securities Australia Itd	Bus	174	2 8	ording.
2-Jun-14	UBS Securities Australia Ltd	Biv	174	α	Ordinary
2-Jun-14	UBS Securities Australia Itd	28.5	174	0.00	Ordinary
2-Jun-14	LIBS Securities Australia Ltd	8.10	VZI	2 0	Ordinary.
2.4 m-14	I IBC Counties Australia 14d	62.6	174	2 0	or all all
2-1-m-14	1 IDC Constitut Australia Ltd	ood o	1/-	0 9	Collingia
l m 1/	I IDO Committee Americal Little	900	***	0 0	Oddinary Codinary
2-1-m-14	I IBC Cognition Australia ted	bouy B	1 010	0 0	Ordinary
2 1 in 14	1 lbc counties Australia Ltd	and a	ZSD, I	50.5	Orginary Orginary
# ;	Obs Securities Australia Ltd	Buy	1/4	8	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	174	18	Ordinary
#1-IIDC-7	Obs Securities Australia Ltd	Buy	84	,	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	1/4	82	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	3,94/	409	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	250	57	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	2,663	276	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	589	61	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	40,655	4,213	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	579	09	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buv	637	99	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	901	E	Ordinary
2-lim-14	ILIBS Securities Australia I to	Bin	10.654	1 104	Ordinan
2-lin-14	II IRS Securities Australia 14d	Bin	2 805	000	Ordinary
11	I IRC Committee Ametralia 14d	600	275	86	O'd'iliai
2-1 m-14	I IRS Securities Australia 144	600	400	3 5	Ordinary
2 Lun 44	Cop Securities Australia Ltd	Buy	492	-0.00	Ordinary
1 1 1	Obs Securities Australia Lid	ng	2,695	DOS.	Ordinary
- Inc-7	Obs Seculities Australia Ltd	any	4,150	430	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	18,663	1,934	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	5,028	521	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	1,612	167	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	6,871	712	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	12,149	1,259	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buv	5.539	574	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Buv	4.150	430	Ordinary
-fin-14	I IRS Securities Australia 14d		13,047	1 35.7	Ording
	The Committee A section and	(ng	13,047	200'1	Countary
# !	Obsideration Australia Ltd	Buy	8,994	337	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	4,150	430	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	14,243	1,476	Ordinary
2-Jun-14	UBS Securities Australia Ltd	Buy	77	8	Ordinary
12-Jun-14	UBS Securities Australia Ltd	Buy	950.3	102	Ordinan
			070,0	76	Currary
-In-14	It IBS Securities Australia Ltd	Buv	4.150	430	Ordinary

Ordinary	Ordinary	Ordinary	Ordinary	nary	nary	nary	nary	Ordinary	nary	nany	nary	nary	Vary	nary	nary	nary	Jary	nany	nary	vac	yary	nary	Jany	Jary	Jary	yary	hany	hary	yery	Jary	yary	ar i	yany	Jary	Jary	hary	Yaer	lary Narv	Jary	hary	Jary	yary	Jary	lary	lany	Z Z	Jav	yary	Jary	yary	Jary	any						
L														54 Ordinary					L	22 Ordinary			00 Ordinary	1						6 Ordinary		Ordinary	L				18 Ordinary	L					3 Ordinary		18 Ordinary	L			8 Ordinary		3 Ordinary	18 Ordinary		18 Ordinary				Ordinary
4,521	4,244	4,521	4,521	4,5	5,075	10,000	10,000	3,307	5,3	3,3(3,399	4,582	3,813	16,154	5.	2,9,	1,976	N O V	4.04	1,122	2,782	òo	1,000	4	10,044	1 000					,	3.761	4,3	2,21	4,0	215	8- 00	132	2,87	10,368	1,630	430				12	1						2					-
43,628	40,955	43,628	43,628	43,628	48,974	96,500	96,500	31,913	51,772	32,492	32,800	44,216	36,795	155,886	2,065	28,390	19,068	38 716	39,420	10,827	26,846	7,855	9,650	4,150	102,715	09 6	19	97	97	154	10 10	36.294	42,171	21,327	39,517	2,075	1/4	1,274	27,483	100,001	15,730	4,150	1771	1,004	174	1,216	174	61	174	174	29	174	203	174	174	174	174	1741
Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Ber	Buy	Buy	Buy	Buy	Buy	Buy	Dirik	Buv	Buy	80,00	Buy	Ruy	Buy	Buy	Buy	Buy	Buy	-																						
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	DBS Securities Australia tod	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Itd	JBS Securities Australia Etd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	Man all all and a second of the land
12-Jun-14		٦	┪	12-Jun-14		┪	T	П	-1	7-Jun-14	- 1	Ī	1	T	12-Jun-14	1	12-Jun-14	Ť	1				7	12-Jun-14	T	Ť	Ī	Ť			12-Jun-14	1	1		12-Jun-14	- 1	2-Jun-14	ı	H	2-Jun-14	- 1	1	12-Jun-14	T	Ī	Ē		П	┪	Ť	Ť	2-Jun-14	Ī	Ī	Ī		- 1	2-lun-14

15.4 m. 14	I IBC Con rition Ametralia 14d			***	Cranar
6-Jun-14	UBS Securities Australia 11d	Buy	ST 1.7.	403	
16-Jun-14	UBS Securities Australia Ltd	Btiv	940 121	95 967	Ordinan
16.1 m-14	I IBC Securities Australia 14d	100	100,000	700,000	Ciginal C
1 1 1 1	one de district à rail	100	312,305	(31,/80)	Ordinary
- 1-11-14	Obsisted Australia Ltd	Sell	11,993	(1,221)	Ordinary
10-Juli-14	ODS Global Asset Ivianagement	Seil	1,895	(193)	Ordinary
6-Jun-14	UBS Wealth Management Australia Ltd	Sell	24,161	(2,457)	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Sell	23,548	(2,398)	Ordinary
7-Jun-14	UBS AG London Branch	Sell	287,187	(59,699)	Ordinary
7-Jun-14	UBS AG London Branch	Buy	288,057	29,699	Ordinary
17-Jun-14	UBS Securities Australia Ltd	Stock borrowed	NAN.	36.970	Ordinan
7-Jun-14	UBS Securities Australia Ltd	a5	5.472	(566)	Ordinary
7-Jun-14	IUBS Securities Australia Itd	2	1 884	100	original or
7 Lin 14	I IDC Commission Assemble 14-4	600	100,1	190	Comilary
7 1:12	Obs Securities Australia Ltd	and .	756,82	2/8/2	Ordinary
/-JUI-14	Obs Securities Australia Ltd	Sell	234,064	(24,236)	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Buy	288,057	29,699	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	62,728	6,504	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	210,428	21,740	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	61,395	(6,307)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buv	76.351	7.854	Ordinary
7-Jun-14	UBS Global Asset Management	Buy	294 780	30.162	Ordinary
7-liin-14	I IBC Wealth Management Australia Ltd	Coll	16.304	(1,000)	S C
7 lun 14	11 DC Woolth Management Australia 1-2	200	10707	(/00'1)	Ordinary
1 - 100-1	Obs. Wealth Management Australia Ltd	, no	576'5	500	Ordinary
7-Jun-14	Obsignation Management Australia Ltd	Ne II	12,083	(1,243)	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Sell	42,617	(4,362)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	274	(28)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	11,548	(1,182)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	01	(1)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	2.061	(110)	Ordinan
7-Jun-14	UBS Securities Australia Ltd	Seil	9.713	(5/0)	Ordinan
7-11 m-14	I BS Sequities Australia 14d	100	00	(0)	i i
7 1 14	Did planting year load coo	Jac I	20	(7)	Ordinary
7 1.17 14	UBS Securities Australia Lid) all	112,169	(11,481)	Ordinary
+1-IIDC-/	Obs securities Australia Ltd	Sell	37,566	(3,845)	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Sell	1,231	(126)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	10	(1)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	10,239	(1,048)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	186	(19)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	29	8	Ordinan
7-Jun-14	UBS Securities Australia Ltd	ll dy	1	5	Ordinan
7-11 m-14	I IDO Comition Australia I to			()	200
100	COO Deconities Australia Liu	II SC	0	Ξ	Ordinary
7-Jun-14	UBS Securties Australia Ltd	Sell	7,943	(813)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	52	(3)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	29	6	Ordinary
7-iiin-14	I BS Securities Australia I to	9	1	1	Ording.
7 1.15 14	TOC CO. Cristian Australia Lat.	= -	2 .	3	O Claraty
/-JUI-14	USS Securities Australia Ltd]ell	10	Ê	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	244,250	(25,000)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Sell	51,586	(2,280)	Ordinary
7-Jun-14	UBS Securities Australia Ltd		29	<u>(E)</u>	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Ruy	1 197	121	Ordinary
7-hin-14	LIBS Securities Australia Ltd	Biray	7 238	750	Ordinan
7-1-10-14	1 IDC Cocumition Aurenalia 144	62.0	552,7	25,	2
1 1 1 1	COO Securities Australia Ltd	ing	1,332	138	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Buy	9,910	1,028	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Buy	7,993	830	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	790	82	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	847	88	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buv	433	45	Ordinary
7-lun-14	LIBS Securities Australia Ltd	Brix	700	ç	Codio
7-tim-14	I IBC Constitute Australia I to	653	1000	77.	Cidiraly
1 - 201	Opp Securities Australia Ltd	pay	9,197	556	Ordinary
1/-Jun-14	UBS Securities Australia Ltd	Buy	8,503	883	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	9,871	1,025	Ordinary
	UBS Securities Australia Ltd	Buy	109'6	966	Ordinary
17-Jun-14	UBS Securities Australia Ltd	Buv	3 596	373	Ordinary
•					A STATE OF THE PARTY OF THE PAR
	II IBC Con writing Australia I to	D	900 9	500	2 1
ı	UBS Securities Australia Ltd	Buy	900'9	623	Ordinary

12-bur 14 USS Securities Australia Ltd Buy Buy 12-bur 14	1	18 Ordinary 21 Ordinary 101 Ordinary
UBS Securities Australia Ltd UBS Securities UBS UBS	1	Ц
UBS Securities Australia Ltd UBS Securities Australia Ltd		L
UBS Securities Australia Ltd UBS Securities Australia Ltd		
UBS Securities Australia Ltd UBS Securities Australia Ltd		7 Ordinary
UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities Australia Ltd UBS Securities UBS UBS UBS UBABA Securities UBS UBS UBBA UBBABA UBBABABA		54 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		89 Ordinary
UBS Securities Australia Ltd UBS Securities UBS UBS UBS UBS UBS UBS UBS UBS UBS UBS		639 Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd U		L
1 UBS Securities Australia Ltd 1 UBS Securities Australia Ltd	492	
UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1,6	L
UBS Securities Australia Ltd UBS Securities Australia Ltd		
UBS Securities Australia Ltd UBS Securities Australia Ltd		L
UBS Securities Australia Ltd UBS Securities UBS UBS UBS UBS UBS UBS UBS UBS UBS UBS		L
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd	222	Oct.
UBS Securities Australia Ltd UBS Securities Australia Ltd		
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd		1
UBS Securities Australia Ltd UBS Securities Australia Ltd	1	34 Ordinary
UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities UBS UBS UB Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Scorities Australia Ltd		
UBS Securities Australia Ltd UBS Securities Australia Ltd		
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd	2,268 235	35 Ordinary
UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities Australia Ltd		
UBS Securities Australia Ltd UBS Securities Australia Ltd		
1.85 Securities Australia Ltd 1.85 S		900 Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd		
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Scorities Australia Ltd	174	18 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		L
UBS Securities Australia Ltd UBS Securities Australia Ltd	9.930 1.029	Ordinary
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Securities Australia Ltd		L
UBS Securities Australia Ltd UBS Securities Lyd UBS Securities		Ordinan
UBS Securities Australia Ltd UBS Securities Australia Ltd		10F Ordinary
UBS Securities Australia Ltd UBS Securities Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management		
UBS Securities Australia Ltd UBS Securities Australia Ltd		
UBS Securities Australia Ltd UBS Goldal Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management		
UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd	5,356 55	555 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		430 Ordinary
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Scorities Australia Ltd		16 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	15,894	
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management	174	18 Ordinary
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management	174	8 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		Ordinos.
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management		
UBS Securities Australia Ltd UBS Securities Australia Ltd	22 056.73	242 Olumary
UBS Securities Australia Ltd UBS Securities Australia Ltd	0,0	1
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management	1,023	
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Social Asset Management UBS Social Asset Management	1,033	77 Ordinary
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Scourities Australia Ltd	42,450 4,39	
UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management		
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Social Asset Management UBS Social Asset Management	59,077 (6,059)	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management	650'9 22'0'65	59 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management		
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS School Asset Management UBS School Asset Management	161,305 16,450	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management	119,312 (12,168)	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management		
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Golbal Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS School Asset Management UBS School Asset Management UBS School Asset Management UBS School Asset Management		
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Golbal Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Scorrities Australia Ltd	(5.000)	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management		
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Clobal Asset Management UBS Securities Australia Ltd		
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Scourities Australia Ird	85 794 8 748	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Clobal Asset Management UBS Clobal Asset Management UBS Clobal Asset Management	,	
UBS Securities Australia Lid UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management IBS Securities Australia It A		
UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management UBS Capacities Australia 1 In	652 773 (66, 683)	L
UBS Global Asset Management UBS Global Asset Management UBS Global Asset Management		
UBS Global Asset Management	211 720 21 576	
UBS Global Asset Management	7	
IN A COLUMNIA DI CITATION TO		
ODS Securities Australia Ltd		
16-Jun-14 UBS Securities Australia Ltd Buy	715,892 72,754	4 Ordinary
UBS Securities Australia Ltd		
Coo Seculities Australia Liu	4,554	Orginary

/ Jun-14	Contract Con	, ,			
7 him 14	Upo Securities Australia Ltd	Buy B.::	74/	8/	Ordinary
7-lin-14	LIRS Sequifies Australia Ind	and a	4,307	312	Ordinary
7 1.00 1.4	Obsideration Augustin Ltd	buy.	6,125	040	Ordinary
7 1 54	ODS Securities Australia Ltd	gnà	3,62	1,019	Ordinary
11-100	Obs Securities Australia Liu	any	107	17	Ordinary
7 7.00	ODS Securities Australia Liu	buy	2,323	243	Ordinary
- Juli-14	UBS Securities Australia Ltd	Buy	5/0/2	217	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Buy	3,509	367	Ordinary
7- Jun-14	UBS Securities Australia Ltd	Buy	1,004	105	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Buy	6/6(1	202	Ordinary
41-Un-	UBS Securities Australia Ltd	Buy	1,042	109	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Buy	9,857	1,030	Ordinary
/-Jun-14	UBS Securities Australia Ltd	Buy	6,658	692	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	6,658	692	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	10,151	1,053	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	10,151	1,053	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	16,889	1,752	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	13,612	1,412	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	347	36	Ordinary
7-Jun-14	UBS Securities Australia Ltd	Buy	321,224	33,322	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Stock returned	N.A	(10,774)	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Stock returned	N/A	(25,260)	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Sell	1,453	(150)	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	22,149	2,280	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	53,351	5,505	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Sell	94,603	(9,762)	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	110,416	11,371	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Sell	1,728,915	(177,958)	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	1,272,284	130,981	Ordinary
-Jun-14	UBS Global Asset Management	Sell	440,891	(45,300)	Ordinary
18-Jun-14	UBS Wealth Management Australia Ltd	Sell	28,181	(2,905)	Ordinary
18-Jun-14	UBS Wealth Management Australia Ltd	Sell	23,379	(2,410)	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	112,617	11,622	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	24,002	2,477	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	73,983	7,635	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	176,862	18,252	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	13,178	1,360	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	23,188	2,393	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	86,018	8,877	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	147,928	15,266	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	287,299	29,649	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	192,802	19,897	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	87	6	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	398.482	41 123	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	49 942	5 154	Critical
18-lun-14	LIBS Securities Australia Ltd	N. B.	48	7,13	or district
8-Jun-14	UBS Securities Australia Ltd	Buy	2	, -	Ordinary
8-lun-14	UBS Securities Australia I td	Riv	3 430	354	Ordinan
18-lun-14	LIBS Sequities Australia I td	Bin	8099	100	Ordinary
18-lun-14	LBS Securities Australia 11d	N 100	15.388	1 588	Ordinary
8-lun-14	I IRS Securities Australia 11d	Bry	900 0	cuc	Ordinary
8-lin-14	I BC Countries Australia 14d	Bury	1 231	305	Ordinary
8-lin-14	1 BC Countries Australia 114	buy.	122,1	97	Ordinary
8.lm-14	I IBS Societies Australia I to	Buny	1,208	100	Ordingry
18-hip.14	I IRS Societies Australia 14d	Buy	505,1	75	Ordinary
-in-14	I IRS Securities Australia 14d	Billy	2 413	0,00	Croinary
18-bin-14	I IRS Societies Australia I I d	Ban.	2,47	54.5	Ordinary
8-lim-14	LIBS Societies Australia I to	Buy	5,014 66,105	5,62	Ordinary
0 tun 14	Liber Control Action Action	bay.	88,103	779'0	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buly	2,419	456	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	2,277	235	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	48	2	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	2,151	222	Ordinary
18-JUN-14	UBS Securities Australia Ltd	8uy	2,180	225	Ordinary

Ordinary	Ordinary	Orginary	Ordinary	Crainary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary																		
979	717	3	63	1,019	1,019	666	566	518	518	1,006	546	532	532	456	1	55	538	1,032	582	643	282	357	700,1	62/	533	1.031	238	1,007	640	640	1,007	1 010	829	1,030	13	1,756	324	163	124	295	5,726	280	833	7,697	1,225	833	280	799	780	202	239	77	13	640	918	832	31	10,000	511	833	78
9,428	6,85/	190'8	/09	9,813	862'6	6,553	6,523	4,962	4,957	6,637	5,231	260'5	260'5	4,368	10	527	5,154	9,887	5,576	6,160	9/5/6	3,374	720,5	644	5.106	9.877	2,280	9,647	6,131	6,131	9,647	9,304	9,762	9,878	125	16,858	3,110	1,565	1.190	5,443	54,826	2,681	9/6'/	73,699	11,729	7,976	2,681	7,646	189'7	1,934	2,000	737	124	6,125	8,785	2,966	297	95,750	4,893	7,976	747
8uy	euy.	śng	Buy	Виу	Buy	ing.	áng.	Buy	Buv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buck	Buy	Buy	Buy	Buy	Ruy	Bliv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	gny	Buy	Buy	Buv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy																
UBS Securities Australia Ltd	UBS Securities Australia Ltd	IDS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	USS Securities Australia LTd	UBS Societies Australia (1)	IRS Societies Australia 14d	UBS Securities Australia Ltd	Securities	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	IBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	IBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	BS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd
17 1.0	Ť	17 1.0 14	T	٦	٦	_	17-Jun-14	- 1		ı	- 1			_	17-Jun-14 U	٦		Ī	17-Jun-14 U	T	Ť	T	T	17-Jun-14 U	Ť	17-Jun-14 U	Ī		- 1	17-Jun-14 U		1	T			17-Jun-14 Ui	- 1	17-Jun-14 U	T	1	ı	l		_	_	17-Jun-14	7	Ť	Ť	- 1	-	17-Jun-14 U	1	Ī	П		1	T	T	T	17-Jun-14 UE

8-Jun-14			-		
	UBS Securities Australia Ltd	Buv		0	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	1,085	112	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	6,783	700	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Виу	1.085	112	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	11.667	1.204	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	3.159	326	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	4,099	423	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	138.741	14.318	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	2,393	247	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buv	0889	711	Ordinary
18-Jun-14	UBS Securities Australia I to	Bliv	3.411	25.	Ordinary
-lim-14	I BC Securities Australia Ltd	Bin	500	100	
18-lim-14	1 IDC Commission Americal and	, ind	203	17	Countary
1 1 1	Opp Securities Australia Lto	hoo	0,441	500	Ordinary
4	UBS Securities Australia Ltd	Buy	14,528	1,500	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	6,441	999	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	53,266	5,497	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	2,403	248	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	3,876	400	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buv	7,268	750	Ordinary
8-lin-14	I IBS Securities Australia 14d	Bray	707	2 5	Codingo
8-1-17 14-17-17	LIPO Countries Australia Ltd	(F)	1,100	5	Ciditaly
14	Tipo Consider Accessing to a	900	9,314	740	O'U'' aly
9-Jun-14	Obs securities Australia Ltd	BUY	196/6	1,028	Ordinary
-JUN-14	UBS Securities Australia Ltd	Buy	6,441	999	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	1,085	112	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	74,846	7,724	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	30.601	21.5	Ordinary
18-hm-14	LIBS Securities Australia Ltd	80.5	3 150	375	Ording
ı	I IBC Congrition Australia Ltd	6.0	22.450	0.7.0	o o
ı	Libe Convision Automotive and		004,03	024,2	(S)
П	Liber Carriers Australia Liu) no	2///9	4000	Ordinary
١	Obs securities Australia Ltd	Buy	3,159	325	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	5,320	549	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	15,078	1,556	Ordinary
	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
	UBS Securities Australia Ltd	Buy	10,979	1,133	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buv	3.159	326	Ordinary
ı	UBS Securities Australia Ltd	`≥₫	80,427	300	Ordinan
ı	LIBC Con retion Australia (+d	, n	1,000	25.	V P
L	COS Securities Australia Liu	ng 2	590,1	7	Ordinary
-	UBS Securities Australia Ltd	Buy	87	6	Ordinary
-	UBS Securities Australia Ltd	Buy	3,159	326	Ordinary
	UBS Securities Australia Ltd	Buy	4,012	414	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	2.975	307	Ordinary
ı	URS Securities Australia I to	N I I	5,310	802	Ordinan
П	District Controls Control C	Say.	010,0	9	Colliary
1	OBS Securities Australia Ltd	suy	008,880	20,000	Ordinary
- 1	UBS Securities Australia Ltd	Buy	97	10	Ordinary
	UBS Securities Australia Ltd	Buy	7,016	724	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	15,533	1,603	Ordinary
	UBS Securities Australia Ltd	Buy	2,355	243	Ordinary
ı	UBS Securities Australia Ltd	Birv	3 983	411	Ordinary
8-inn-14	I IRS Societies Australia 11d	P. IV	34 306	0000	, in the second
100	Con Securities Australia Lid	Buy	007,46	000,0	Ordinary
0-Juli-14	Obs securities Australia Ltd	pay	3,159	976	Crainary
-Jun-14	UBS Securities Australia Ltd	Buy	8,073	834	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	69,269	6,846	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	2,197	722	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	29	٣	Ordinan
8-lun-14	I BS Securities Australia Ltd	B. S.	6316	300	or din so
14	IDS Committee Australia 14d	bod.	2,100	077	Oldlinary
+1-1mr-01	Obs securities Australia Ltd	gng	,	×	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	2,168	224	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
	UBS Securities Australia Ltd	Buy	89	7	Ordinary
	UBS Securities Australia Ltd	Buy	2,168	224	Ordinary
L					
	The contract of the contract o	2.0	0,	-	, included

Г	1	Т	ī	1	_						_		_	_	_			-	_	_	т	1		-	_	_	7	<u> </u>		_	_	_	1 1			_	 	_	-	_	_		_	_	т-	-T	_	_	_	_	_		, , , , ,			_	_	_
in C	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Citationy
1 103	1,103	5 113	226	98	866'9	226	526	222	337	220	229	722	173	668	1,260	-	6,583	280	492	328	12	16	400	173	2 101	5,10	118	6,981	52	1,533	326	7,159	326	326	326	9,142	821	400	2,524	491	326	326	181	548	5,5/1	331	326	326	326	503	326	796	484	1,356	1,028	7	326	1 177
00701	16,066	575'67	2,190	833	118/29	2,190	2,190	2,151	3,266	5,523	2,219	2,200	1,676	8,711	12,209	10	63,789	5,620	4,767	2 150	116	155	3,876	1,676	20.02	10,05	1,143	67,646	204	14,855	3.159	69,371	3,159	3,159	3,159	88,586	7,955	3,876	19 583	4.758	3,159	3,159	1,754	5,310	54,003	3.207	3,159	3,159	3,159	4,8/4	3 159	7,713	4,690	13,140	9,961	03.50	3,159	7,1,1
A la	Buy	Buv	Buy	Binc	8uv	Buy	Buy	Buy	Buy	Bliv	Buy	Buy	Buy	Buy Brit	Buy	Buy	Buy	Buy	Bliv	Buy	Buy.	Buy	Buy	Buy	Buy	any	Bay	Buy	Buy	Buy	Buy	anà	Buy	inal																								
BS Securities Australia Ital	UBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	DBS Securities Australia Ltd	LIRS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	IRS Securities Australia Ltd	JBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	USS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	105 Securities Australia Ltd	UBS Securities Australia Ltd	אם אברתונוכה שמהוומות בנה
Г	18-Jun-14	1	ı		l.					٦		П		18-Jun-14	П		٦	T	18-Jun-14	1	18-Jun-14			T	18-11-14	T	Ť		1	18-Jun-14	۳	Ť	П	18-Jun-14	T	Ť	П	- 1	18-Jun-14		Ť	Ť		1	18-lun-14	T	Ī	Ħ	_	Ť	18-lun-14	Г	П	Ť	Ť	Ť	18-Jun-14	

	UBS Securities Australia III	202	0.		Ordinary
8 In 14	I IDC Cocumision A costralia 1+4	7.0	007.0		
10-101-14	Ups Securities Australia Ltd	an)	b,428	564	Ordinary
1	Obs securities Australia Lid	Buy	4,888	505	Ordinary
-un01	UBS Securities Australia Ltd	gon	10,425	1,077	Ordinary
-Jun - 14	Ubs securities Australia Ltd	ROY	4,385	453	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Buy	1,946	201	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	13,804	1,426	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	11,161	1,153	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	1,084	112	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	6,437	999	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	1,084	112	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buv	3.156	326	Ordinary
F-Jun-14	UBS Securities Australia Ltd	Buv	6.331	654	Ordinary
18-Jun-14	URS Securities Australia Ltd	Bin	3.156	300	Ordinary
18-Jun-14	LIRS Securities Australia 14d	, and	4000	777	Ordinary
8-hin-14	LIBS Securities Australia Ltd	, and	200, 0	4 4	Ordinary
1 1 1 0	Opp Securities Australia Liu	, and	3,465	ngs	Ordinary
#1-1m-0	Obs Securities Australia Ltd	Buy	542	95	Ordinary
101	Upo Securities Australia Ltd	Buy	//4	28	Ordinary
101	Upo Securities Australia Ltd	Buy	16,191	1,466	Ordinary
10-Jun-14	UBS Securities Australia Ltd	gnà	7,018	725	Ordinary
5-70U-14	UBS Securities Australia Ltd	Buy	2,178	225	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	8,247	852	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	17,705	1,829	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	116	12	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	136	14	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	93,673	2/29'6	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	2.188	226	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	134 000	13.843	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3 553	367	Ordinary
8-Jun-14	UBS Sequrities Australia Itd	Biv	791 6	777) drain o
8-1 in-14	I IRS Securities Australia 14d	Bony	1 326	100	ording)
2-lim-14	I IRS Securities Australia 14d	6.00	726.23	000	Ordinary
1 1 0	The Committee A section of the contract of the	(ha	/oc'oc	2,025	Clainary
-100-14	Obsider Australia Ltd	Buly	079	64	Ordinary
6-Juli-14	Ubs securities Australia Ltd	Buy	24,636	2,545	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Buy	2,459	254	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	532	22	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	3,494	361	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	2,120	219	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	4.888	505	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	271	28	Ordinary
8-lin-14	LIBC Constitute Assets and and and and and and and and and and	D. C.	1,7	07	O'dillialy
1 1 1 1	ODS Securities Australia Ltd	Áng	9/0/90	56/3	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	17,927	1,852	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	2,778	287	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	2,226	230	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	62.368	6.443	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buv	065	19	Ordinary
8-lim-14	BS Securities Australia Ltd	Brix	3 223	222	Coling
14	I IBS Securities Australia I to	Birk	6 705	502	Ordinary
18-lim 14	I IDO Consistent Australia Lan		0,150	20/	Gulliary
1	Obs Securities Australia Ltd	Buy	601,2	577	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	1,2/8	132	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,620	374	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	203	21	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	18,411	1,902	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buv	3.891	402	Ordinary
8-Jun-14	UBS Securities Australia I to	Buy	2 149	222	Ordinan
14	Late of section A resistance 2 2011		67,12	777	, G
1100	Obsideration Australia Liu	bug	2,108	5 77	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Buy	2,159	223	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	310	32	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	2,207	228	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	2,207	228	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	2,207	228	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	2.178	225	Ordinary
8-Jun-14	UBS Securities Australia I td	Buy	2 168	224	Ordinary
			200		
18-110-17	LIBS Spainting Australia Ltd	P. o.	32025	1 000	in in

18-Jun-14 18-Jun-14 18-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
Jun-14 Jun-14			2		
-Jun-14	UBS Securities Australia Lid	80	29	٣	Ordinary
	UBS Securities Australia Ltd	Buy	14.520	1.500	Ordinary
18-Jun-14	UBS Securities Australia I td	Buy	4 075		Ordinary
8-tun-14	UBS Securities Australia 1td	Riv	207.3	124	Ordinary
14	I IBS Securities Australia I to	Buch	30.10		Ordinary
18-117	I IRC Congritise Australia 14d	Pun	140		Ordinary
8-1-0-14	IBS Societies Australia 14d	and a	130 66	0,10	Ordinary
11010	I IDC Commission Australia Lea	9	100,00		Ordinary
1	The same same same	gng	35		Ordinary
9-7017-14	ODS Securities Australia Liu	ána	1,084	711	Orginary
Jun-14	UBS Securities Australia Ltd	Buy	6,979	721	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	5,973	617	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	13,468	1,392	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	38 700		Ordinary
8.lim.14	I IBS Socirities Australia I to	Bank	11 060		College
1	UBC Contribion Australia 144	(na	000,11	t+	Countary
t -	Obs Securities Australia Liu	eu.	3,6/2		Ordinary
П	ODS Securities Australia Lid	Ang	6,224	643	Ordinary
1	UBS Securities Australia Ltd	Buy	8/		Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	166,196	17,	Ordinary
- 1	UBS Securities Australia Ltd	Buy	5,305		Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	110,662	11,432	Ordinary
- 1	UBS Securities Australia Ltd	Buy	136	14	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	52	e	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
Jun-14	UBS Securities Australia Ltd	Suv.	3.156	326	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Riv	3 156	325	Ordinary
8-110-14	LIRS Securities Australia 14d	Buy	2156	326	Ordinan
1, 0	DO Complete August 144		22.10	220	Colliary
1 1 1	Obs Securities Australia Liu	ond	3,150	320	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
- 1	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
l	UBS Securities Australia Ltd	Buy	6,302	651	Ordinary
ı	UBS Securities Australia Ltd	Buv	73 355	7 578	Ordinary
8-11n-14	185 Countities Australia Ltd	100	1000	C11.C	Cadina
1	ODS Securities Australia Ltd	had	30,124	3,112	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	4,704	486	Ordinary
- 1	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	15,527	1,604	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	196'6	1,028	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	61	^	Ordinary
0 lun 1.4	I IOO Countries Australia I to	500	2166	300	Ordinary
1-In	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	7,473	772	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	3,223	333	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	42.747	4416	Ordinary
8-lin-14	IBC Securities Australia Hol	100	7L / 33	101	Ordingry
	ממס ספרתו וונפס שתפון מוומ בית	bod)	0/4/00	167,0	Cidinary
10-JUL-14	Obsisecurities Australia Ltd	Buy	691,82	018,2	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	23,435	2,421	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	3.156	326	Ordinary
18-lin-14	IBS Securities Australia I1d	Bit	34 567	3 571	Ordinan
		62	OC'ho	1000	Ordinally Control
8-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buv	14.617	1.510	Ordinary
8-1 m-14	IBS Securities Australia 144	200	2 156	300	Ordingo
	DO Committee Australia ted	600	5,100	000	(in in in in in in in in in in in in in
8-Jun-14	UBS Securities Australia Ltd	Buy	156,9	1,028	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	22,574	2,332	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	5.305	548	Ordinary
Γ	BS Securities Australia Ltd	812	6 137	769	Ordinary
T	DC Counties Australia 144	600	0,127	100	Countries
10-1011-14	DOS SECULITIES AUSTIGNIA LICA	ng	3,130	370	Orginary
T	UBS Securities Australia Ltd	Buy	5,740	593	Ordinary
П	UBS Securities Australia Ltd	Buy	75,727	7,823	Ordinary
	IBS Societies Australia Ltd	Rick	070 30	102.0	

18-10-14 UBS Securities Australia Ltd Buy	1,259	
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,259	
UBS Securities Australia Ltd UBS Securities UBS Securiti		130 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,259	
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,259	130 Ordinary
UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities Australia Ltd		7,203 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	4,145	428 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	56.783	L
UBS Securities Australia Ltd UBS Securities Australia Ltd	200	60 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5.482	
UBS Securities Australia Ltd UBS Securities Australia Ltd	1 260	
UBS Securities Australia Ltd UBS Securities Australia Ltd	100 0	
UBS Securities Australia Ltd UBS Securities Australia Ltd	7/E CDa'c 6r	
UBS Securities Australia Ltd UBS Securities UBS Securities UBS UBS Securities UBS UBS Securities UBS UBS Securities UBS UBS Securities UBS UBS Securities UBS UBS Securities UBS UBS Securities UBS UBS Securities UBS UBS Securities UBS Securities UBS Securities UBS Securities UBS UBS Securities UBS UBS Securities UBS Securities UBS Securities UBS UBS Securities UBS UBS Securities UBS UBS Securities UBS Securities UBS UBS Securities UBS UBS Securities UBS Securities UBS UBS UBS UBS UBS UBS UBS UBS UBS UBS	0,444	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	678	
UBS Securities Australia Ltd UBS Securities UBS Secu	006'96	000 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		400 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		266 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	32,212	3,326 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3.673	
UBS Securities Australia Ltd UBS Securities Australia Ltd	33.878	3.498 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5.485	
UBS Securies Australia Ltd UBS Securies Australia Ltd UBS Securies Australia Ltd UBS Securies Australia Ltd UBS Securies Australia Ltd UBS Securities Australia Ltd	5 640	
UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities Australia Ltd UBS Securities UBS S	28 000	
UBS Securities Australia Ltd UBS Securities Australia Ltd		
UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities Australia Ltd UBS Securities UBS S	0,444	
UBS Securities Australia Ltd UBS Securities Australia Ltd	34,953	
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,012	
UBS Securities Australia Ltd UBS Securities Australia Ltd	24,971	77 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	35,060	20 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,260	130 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	78.915	
UBS Securities Australia Ltd UBS Securities Australia Ltd	23.266	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Lustralia Ltd UBS Securities Australia Ltd UBS Securities Lustralia Ltd UBS Securities Australia Ltd	202,02	in in in
UBS Securities Australia Ltd UBS Securities Australia Ltd	###*O	
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,485	See Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,492	
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,519	60 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Ltd	2,965	306 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	626	101 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	7,917	
UBS Securities Australia Ltd UBS Securities Australia Ltd	2 907	SOO Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	00 456	
UBS Securities Australia Ltd UBS Securities Australia Ltd	200	l
UBS Securities Australia Ltd UBS Securities Australia Ltd	100	
UBS Securities Australia Ltd UBS Securities Australia Ltd	058,5	
UBS Securities Australia Ltd UBS Securities Australia Ltd	96,900	
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,537	365 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,876	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	4 803	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	417	L
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	/14	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	7,936	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	67,316	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	95,542	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	869'08	L
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	8.333	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	17.083	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	038 C	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	6,007	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	4,050 418	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	16,880	42 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	4,041	
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd IBS Securities Australia Ltd	24,593	38 Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd		L
UBS Securities Australia Ltd UBS Securities Australia Ltd I RS Securities Australia Ltd	300	
UBS Securities Australia Ltd URS Securities Australia Itd		out of our land
UBS Securities Australia Ltd	0,400	
I IBS Socurities Australia Ltd	4,603	475 Ordinary
מפט מכיפון וונים שמנומוום בנם	1,725	78 Ordinary
18-Jun-14 UBS Securities Australia Ltd Buy	1 689'6	
	4.562	471 Ordinary
	3 148	
Т	0.1401	325 Ordinary

3
Buy Buy Buy Buy Buy Buy Buy
Buy Buy Buy Buy Buy Buy Buy Buy
809 809 809 809 809 809 809 809
Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy
Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy
Buy Buy Buy Buy Buy Buy Buy Buy Buy

10-00	כמים הכינונים שכפים שום דום	200			
18-Jun-14	UBS Securities Australia Ltd	Buy	94,371	9,739	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	8,014	827	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	726	75	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buv	3.653	377	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	36.522	3 769	Ordinary
18.410-14	I IRC Securities Australia Ltd	7.0	12 000	001,0	0.00
18-4 m-14	I BC Countries Australia 144	boy.	12,000	055,1	Ordinary
10 10 14	Upo Comition Australia Ltd	Buy	35,737	3,000	Orginary
1	I lbc committee Australia Ltd	son a	68/'87	1/6/7	Ordinary
1	UBS Securities Australia Ltd	Buy	12,549	1,295	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	55,417	5,719	Ordinary
- 1	UBS Securities Australia Ltd	Buy	5,485	299	Ordinary
- 1	UBS Securities Australia Ltd	Buy	53,295	5,500	Ordinary
١	UBS Securities Australia Ltd	Buy	2,907	300	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	5,058	522	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	17,830	1,840	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buv	2.578	266	Ordinary
18-Jun-14	UBS Securities Australia Ltd	N B	65,892	008.9) raining
18-lin-14	I IRS Securities Australia I to	Bush	50,02 E 48E	0000	ordinary.
10 101	I IDC Commission Assistant Library	bay.	0,400	oon	Orginary
	UDS Securities Australia Ltd	Buy	4,070	470	Ordinary
	Upo Securities Australia Ltd	Buy	14,370	1,483	Ordinary
1	UBS Securities Australia Ltd	Buy	1,231	/71	Ordinary
18-JUN-14	UBS Securities Australia Ltd	auy	9,642	382	Ordinary
-1	UBS Securities Australia Ltd	Buy	3,653	377	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	3,653	377	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	3,653	377	Ordinary
- 1	UBS Securities Australia Ltd	Buy	12,888	1,330	Ordinary
	UBS Securities Australia Ltd	Buy	15,804	1,631	Ordinary
	UBS Securities Australia Ltd	Buy	4,283	442	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	28,760	2,968	Ordinary
	UBS Securities Australia Ltd	Buy	3,876	400	Ordinary
ı	UBS Securities Australia Ltd	Buv	18 905	1 951	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	2 229	230	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	89 710	9.758	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Biv	6 444	599	Ordinary
8-Jun-14	IUBS Securities Australia Ltd	Buy	3 207	331	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buv	5.349	552	Ordinary
18-Jun-14	LBS Securities Australia Ltd	S _I N	6.444	999	Ordinan
8-in-14	LIRS Securities Australia 11d	Bush	2,775	200	Colingo
8-lin-14	II IRS Securities Australia I to	B. (2)	16,007	1 753	(Burgo
10 10	LIDE Commission Australia Lau	(ng	186,01	000	O'Carraiy
1 - In 3	LIDE Countition Australia Ltd	eug B	2,208	977	Orginary
0-1011-14	OBS Securities Australia Ltd	Buy	34,525	5,563	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	18,033	1,861	Ordinary
ı	UBS Securities Australia Ltd	Buy	3,101	320	Ordinary
- 1	UBS Securities Australia Ltd	Buy	89,710	9,258	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	1,996	206	Ordinary
- 1	UBS Securities Australia Ltd	Buy	2,578	266	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	5,310	548	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	5,310	548	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	388	40	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	85,243	8,797	Ordinary
8-Jun-14	UBS Securities Australia Ltd	Buy	24,409	2,519	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buv	18.430	1 902	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	19 787	2.042	Ordinary
18-tim-14	IRS Securities Australia I to	Berry	22,412	2000	Galliany
8-km-14	I IRS Securities Australia I to	Berry	1 182	122	Crdings
8-lin-14	188 Securities Australia 144	Berry	20.05	77	Ordinary Series
18-lin-14	I BS Securities Assertal 14d	Bux	300,0	200	Ordinary
1	LIBS Securities Australia 1td	Buy	56 570	028.9	Ordinary
1	II IRS Securities Aristralia 11d	Brin	A 593	N7.N	Ordinan
L	LIBC Countition Australia 144	, and	CC 7 2	4/4	O G
	I IDC Countries Australia Ltd	áng d	0,4/3	000	Orginary
110000	ODS Securities Australia Ltd	, and	50	4	Ordinary
0-Jun-14	Obs securities Australia Ltd	Buy	11,66/	1,204	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	5,310	548	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	5,688	287	Ordinary
18-jun-14	UBS Securities Australia I to	2.	1/28	Lic	-
	כמום שייים שמשויים בייוו		1.70	00	Ordinary

111
\$ 8 8
BG
Buy
a a
Buy
Buy
<u>\$</u>
2 2
- 1
ı
Γ
T
8
96
βğ
Buy
ń
BG
Buy
BG.
ans.
in a
Ť
Buy
Γ
Buy
Buy
_
Buy
Buy
Buy
Buy
æ
Buv
BUV
BG
Bull
B
T
1
ı
-
ı
1

19-lin-14	II BS Societies Australia Ltd	2 0	215 216	250.15	(in the contract of the contra
n-14	UBS Securities Australia Itd	N. S.	2 181	350	Ordinary
19-Jun-14	UBS Securities Australia Ltd	i es	1,233,867	(175,738)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	BCv	848,639	86,587	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	461.160	(47.013)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Buv	2.463.580	251 455	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	233,390	(23,793)	Ordinary
9-Jun-14	UBS Wealth Management Australia Ltd	Buv	9.404	953	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	164,151	(16,716)	Ordinary
19-jun-14	UBS Securities Australia Ltd	Sell	12,098	(1,232)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Buy	230,259	23,448	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	4,193	(427)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	194,436	(19,800)	Ordinary
n-14	UBS Securities Australia Ltd	Sell	209'59	(6,681)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Buy	1,905	194	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	275,677	(28,073)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	54,707	(1/5/5)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	14,730	(1,500)	Ordinary
9-jun-14	UBS Securities Australia Ltd	Sell	120,619	(12,283)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	1,921	(802)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	93,815	(9,534)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	14,307	(1,454)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	413	(45)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	413	(42)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	403	(41)	Ordinary
un-14	UBS Securities Australia Ltd	Sell	3,434	(349)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	2,362	(240)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	66,793	(8,109)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	17,318	(1,760)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	20,674	(2,101)	Ordinary
un-14	UBS Securities Australia Ltd	Sell	4,989	(207)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	108	(1)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	32,639	(3.317)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	3,582	(364)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	3,149	(320)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	6,002	(610)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	lleS.	9,220	(937)	Ordinary
Jn-14	UBS Securities Australia Ltd	38	1.151	(117)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	lleS.	118	(12)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	les	3.446	(350)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	8.861	(006)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	les	5,513	(290)	Ordinary
h-14	UBS Securities Australia Ltd	38	1.974.000	(200,000)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	ll as	57.033	(5 796)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	3.149	(320)	Ordinary
9-hun-14	UBS Securities Australia I td	la?	08	(070)	Colingo
19-Jun-14	UBS Sequrities Australia I td	la de	133	E V	Coling
14-14	UBS Securities Australia 11d	Sell	433	(44)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	433	(44)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	ll as	493	(50)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	433	(44)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	473	(48)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	433	(44)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	473	(48)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	424	(43)	Ordinary
19-jun-14	UBS Securities Australia Ltd	Sell	433	(44)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	404	(41)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	404	(41)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	404	(41)	Ordinary
ın-14	UBS Securities Australia Ltd	Self	414	(42)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Self	6,885	(669)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	9,850	(1,000)	Ordinary
9-Jun-14	UBS Securities Australia Ltd	Sell	3,152	(320)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	414	(CV)	Ordinan
l				/75	7000
9-Jun-14	UBS Securities Australia Ltd	Jes	414	(42)	Ordinary

	JBS Securities Australia Ltd	Buy	3,924	405	Ordinary
18-Jun-14	UBS Securities Australia Ltd	Buy	7,229		
18-Jun-14	UBS Securities Australia Ltd	Buy	11,638	1,201	
=	UBS Securities Australia Ltd	Buy	9,661		
┪	UBS Securities Australia Ltd	Buy	3,924	405	
7	UBS Securities Australia Ltd	Buy	8,353		Ordinary
7	BS Securities Australia Ltd	Buy	388		
7	UBS Securities Australia Ltd	Buy	4,002		
7	UBS Securities Australia Ltd	Buy	6,502		
7	UBS Securities Australia Ltd	Buy	67,336	6,949	
4	BS Securities Australia Ltd	Buy	23,896		Ordinary
-	IBS Securities Australia Ltd	Buy	1,502		Ordinary
Ħ.	UBS Securities Australia Ltd	Buy	3,876	400	L
~	UBS Securities Australia Ltd	Buy	2,413		
۲	UBS Securities Australia Ltd	Buy	36,454		
۲	UBS Securities Australia Ltd	Buy	75,582	7,800	Ordinary
۲	BS Securities Australia Ltd	Buv	17.859		
۲	BS Securities Australia Itd	Riv	5 310		
+	LIBS Securities Australia Itd	in a	3 470		
4-	I IRC Societities Australia 14d	P. P.	300 3		
Ŧ	BC Counties Australia (14)	, in a	760.0	8	O Company
+	DO Secondary Australia Ltd	p.a.	1070		
۲	PDC Committee Australia Ltd	bay.	231 C		
4-	DS Securities Abstraria Ltd	boy.	5,130	25	
+	bs securities Australia Liu	anà	/8		
4	JBS Securities Australia Ltd	Buy	1,520		
4	UBS Securities Australia Ltd	Buy	4,172	493	Ordinary
4	UBS Securities Australia Ltd	Buy	4,250		
4	BS Securities Australia Ltd	Buy	1,713	177	
쒸	UBS Securities Australia Ltd	Buy	7,744		
-4	UBS Securities Australia Ltd	Buy	5,518		
4	UBS Securities Australia Ltd	Buy	50,839	5,252	
러	JBS Securities Australia Ltd	Buy	14,230		L
۲	BS Securities Australia Ltd	Buy	3,282		L
屵	UBS Securities Australia Ltd	Buy	1,994		
H۲	BS Securities Australia Ltd	Buv	5.305	548	L
1-	UBS Securities Australia I td	Buy	56.899	"	
┾	LIBS Securities Australia Itd	Bits	1268		Ordina
4-	BC Countries Australia 14d	, i d	2002,		Ordinary
4-	Obsideration Australia Liu	buy O	0,013		Orginary
4-	Do Securities Australia Liu	, and	200,0		
4	Obs Securities Australia Ltd	Buy	560'/	99/1	
4	UBS Securities Australia Ltd	Buy	781		
4	BS Securities Australia Ltd	Buy	3,678		Ordinary
-4	BS Securities Australia Ltd	Buy	5,382	556	Ordinary
-2	UBS Securities Australia Ltd	Buy	4,162		Ordinary
2	UBS Securities Australia Ltd	Buy	37,752	3,900	Ordinary
尸	UBS Securities Australia Ltd	Buy	22.477	2.322	Ordinary
尸	UBS Securities Australia Ltd	Buv	22,400	2.314	Ordinary
F	85 Securities Australia I td	Buy	620		Ordinan
1=	UBS Securities Australia I to	Briv	27 665	2 2	Ordinary
#=	JBS Sequities Australia Ltd	Riv	8779		Ordinan
4-	IBS Sequerties Australia I tol	N. W.	23.742	2 401	Ording
7-	UDC Countries Australia Ltd	, and	242,62	2,401	Cidinary
4	55 Securities Australia Ltd	ing.	01	- 00	Ordinary
4	DDS SECURITIES AUStralia Ltd	ing.	4,102		Ordinary
4	UBS Securities Australia Ltd	ans	560/1		Ordinary
7	UBS Securities Australia Ltd	Buy	20,638	2	Ordinary
4	UBS Securities Australia Ltd	Buy	3,562	36	Ordinary
긕	3S Securities Australia Ltd	Buy	77	8	Ordinary
괵	JBS Securities Australia Ltd	Buy	11,287	1,166	Ordinary
괵	3S Securities Australia Ltd	Buy	3,978	411	Ordinary
2	JBS Securities Australia Ltd	Buy	252	56	Ordinary
2	JBS Securities Australia Ltd	Buy	19,980	2,064	Ordinary
₽	JBS Securities Australia Ltd	Sell	1,279		Ordinary
₽	JBS Securities Australia Ltd	Sell	89		Ordinary
₽	JBS Securities Australia Ltd	Sell	39		Ordinary
43	IBS Securities Australia I to	Stock borrowed	AVA	15 71	Ordinary
4	Securities Australia Lid	TOWN DOLLOWS			
2	3S Securities Australia Ltd	Stock borrowed	ΥN		
F	LIBS Securities Australia I to	218	19 803	3000	Ordinan
긱	35 Seculities Australia Ltd	BUJ	C00/61		

UBS Securities Australia Ltd UBS Securities Australia Ltd	Sell Sell Sell Sell Sell Sell Sell Sell	25.248 25.248 25.248 25.248 26.000 42,000 42,000 42,000 58,744 56,925 66,334 775,732 86,935 106,276 11,62,000 12,646 23,647 24,6	(53) (63) (869) (863) (863) (877) (877) (877) (878) (10,948) (10,9	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		25,348 8,551 8,551 3,060 4,000 5,000 5,000 106,276	(537) (869) (817) (917) (925) (10,942)	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		25,948 8,551 3,050 3,050 3,050 61,332 86,972 27,440 106,354 106,354 106,354 106,354 106,354 106,354 106,354 106,354 106,354 106,354 106,356 106,365 10	(1537) (869) (817) (877) (877) (877) (877) (10,953) (10,9	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		8,551 3,050 42,000 42,000 58,743 50,397 10,532 10,532 175,722 175,722 175,722 1,440 10,536 1,440 10,536 1,440 10,536 1,440 1,06,276 1,06,	(869) (4,277) (4,277) (6,251) (6,251) (10,953) (Ordinary Ordinary
1,UBS Securities Australia Ltd UBS Securities Australia Ltd		3,060 42,000 58,740 10,324 10,324 10,534 10,577 10,	(113) (5,982) (6,982) (10,945)	Ordinary Ordinary
1 UBS Securities Australia Ltd 1 UBS Securities Australia Ltd		42,000 58,743 90,997 10,5354 106,276 106,27	(4.27) (5.982) (3.982) (6.251) (6.251) (10.953) (70.9483) (10.9483) (10.9483) (10.9483) (10.9483) (10.9483) (10.9483) (10.9483) (10.9483) (10.9483) (10.263) (10.263) (10.263) (10.000)	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		5,8743 50,937 61,332 86,972 27,440 106,354 106,356 1,480,39 1,116 591,116 591,116 591,116 591,116 91,116 92,813 16,395	(5,982) (6,251) (6,251) (10,953) (10,953) (10,953) (10,953) (10,953) (10,953) (10,953) (10,050) (10,00	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		30,997 61,322 86,972 27,440 106,354 775,722 775,722 775,722 775,722 775,722 706,276 92,313 9,904 30,907 30,807 30,	(3.163) (8.957) (10.828) (79.888) (79.889) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.945) (10.948) (10.9	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		61,322 86,972 27,440 106,254 775,722 106,276 106,276 591,116 92,313 NA 39,904 39,904 3,904 3,904 3,905 2,646 8,876 16,395 9,673 9,673 9,673 18,905 2,646 92,805 92,805 9,673 18,905	(6.25) (8.357) (10.953) (10.953) (10.9483) (10.9483) (10.9483) (10.9487) (9.263) (9.26	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		86,972 27,440 106,354 1,06,356 1,06,376 591,116 591,116 591,116 91,016	(19,983) (10,9889) (10,9889) (10,9889) (11,02,945) (15,024) (10,020) (10,00	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		27,440 106,354 106,276 1,498,039 1,416 591,116 92,313 10,44 39,030 16,836 16,83	(10,343) (10,343) (10,343) (10,343) (10,342) (10,342) (10,342) (10,342) (10,342) (10,000) (10	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		106,354 106,372 106,276 106,275 116,295 20,116 92,313 10,290 2,646 8,376	(10,953) (10,948) (10,948) (154,278) (60,877) (9,507)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securities Australia Ltd		775,722 1,498,016,276 1,416 92,313 NAA 39,904 2,646 8,876 16,395 16,395 16,395 16,395 16,395 16,395 16,395 16,395 16,395 16,395 16,395 16,395 16,395 16,395 16,395 17,395 18,395	(19,889) (10,4278) (15,4278) (60,8777) (9,265) (10,265) (10,265) (10,265) (10,267) (10,000) (Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1085 Securities Australia Ltd 1085 Securities Australia Ltd		106,276 1,498,039 1,416 92,313 10,414 2,646 2,646 18,965 16,93 9,613 9,613 18,948 18,966 18,9	(10,345) (154,278) (60,877) (9,507) (9,507) (9,265) (9,050) (10,00	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1085 Securities Australia Ltd 1085 Securities Australia Ltd		1,498,039 92,313 NA NA 39,904 2,546 8,876 8,876 8,973 9,050 9,050 16,395 9,050 10,395	(154,278) (60,877) (9,267) (9,265) (9,265) (9,265) (90	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		1,20,039 1,111,6 1,03,131 1,00,031 1,00 1,00	(60,877) (9,507) (9,507) (9,507) (9,507) (9,087) (985) (10,000 (10,000) (10,000) (12,745) (15,769) (15,769) (15,769) (15,769) (15,769) (15,769) (15,769) (15,769) (15,769)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1005 Securities Australia Ltd 1005 Securities Australia Ltd		951,115 95,313 96,333 96,33 96,33 96,33 96,33 96,73	(9.507) (9.507) (9.265) (9.265) (270) (905	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1915 Securities Australia Ltd 1915 Securities Australia Ltd		92,313 94,04 34,04 2,646 8,376 16,395 9,673 9,050 9,050 10,305 10,305 10,40	(9,265) (9,265) (2,70) (905) (Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		39.904 2,646 2,646 18.395 16.395 96.73 96.73 92.806 92.806 92.806 92.806 92.806 92.806 92.806 92.806 92.806 92.806 93.806	(9.265) (9.266) (2.701) (905) (985) (985) (985) (987) (9.470) (9.470) (9.470) (15.769) (15.769) (15.769) (16.769) (17.769) (18.769) (19.769) (19.769)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1085 Securities Australia Ltd 1085 Securities Australia Ltd	Buy Sell Sell Buy Buy Buy Buy Sell Sock returned Siock returned Siock returned Siock returned Suy Sell Sell Sell Sell Sell Sell Sell Sel	39,904 2,646 8,8376 16,595 9,673 9,673 9,0	(1270) (905) (905) (905) (905) (905) (985) (985) (985) (987) (15,769) (15,769) (15,769) (15,769) (15,769) (16,769) (17,769) (18,769) (18,769) (18,769) (18,769) (18,769) (19,7	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Sell Sell Buy Buy Sell Buy Sell Siock returned Siock returned Sell Sell Siock returned Sell Sell Sell Sell Sell Sell Sell Se	2,646 8,676 16,395 9,673 9,605 256,686 256,686 2,806 NA NA NA NA NA NA NA NA NA NA NA NA NA	(270) (965) (985) (985) (985) (10,000 (9,470) (15,769) (15,769) (15,769) (15,769) (15,769) (15,769) (15,769) (15,769) (16,769)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Sell Buy Buy Buy Buy Sell Sell Sell Sell Sock returned Stock returned Buy Buy Buy Buy Buy Buy Buy	8.876 16.395 98.050 256.686 92.806 NA NA NA NA NA NA NA NA SPS 888 2.990 2.990 2.990 2.990 2.990 3.893 8.888	(905) 1,673 (985) 10,000 26,173 (9,470) (15,46) (15,769) (15,769) (678) 39,957	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Sell Buy Buy Buy Sell Siock returned Stock returned Stock returned Suck Buy Sell Sell Sell Sell Sell Sell Sell Sel	16,395 9,673 9,673 9,673 92,806 10,4 10,4 10,4 10,4 10,3 10,3 10,3 10,3 10,3 10,3 10,3 10,3	(985) (985) (985) (967) (9,470) (15,445) (15,763) (15,763) (15,763) (16,763) (18,763) (18,763) (19,763)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Self Buy Buy Self Siock returned Siock returned Self Self Self Self Self Self Self Self	9,673 98,050 256,886 92,806 NA NA NA NA 2,990 2,990 6,617 389,848	(985) 10,000 26,173 (9,470) (12,445) (15,769) 306 (678) 39,957	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy Buy Sell Sinck returned Sinck returned Self Suy Self Suy	98,050 256,686 92,806 92,806 NA NA 2,990 2,990 6,617 389,848	(10,000) 10,000 26,173 (9,470) (15,769) 306 (678) 39,957	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy Sell Slock returned Slock returned Buy Sell Suy Sell Suy	256,050 256,066 92,806 N/A N/A 2,990 2,990 6,617 389,848	(15.769) (15.769) (678) (678) 306 (678)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1005 Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd UNS Securities Australia Ltd	Stock returned Stock returned Stock returned Buy Sell Buy	225,086 92,806 N/A N/A 2,990 2,990 6,617	(9,470) (9,470) (15,769) 306 (678) 39,957	Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Sell Stock returned Stock returned Buy Sell Buy Sell Sell Sell Sell Sell Sell Sell Sel	92,806 N/A N/A 2,990 2,990 6,617	(9,470) (2,445) (15,769) 306 (678) 39,957	Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Stock returned Stock returned Buy Sell Buy	N/A N/A 2,990 6,617 389,848	(2,445) (15,769) 306 (678) 39,957	Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Stock returned Buy Sell Buy Buy	2,990 2,990 6,617	(15,769) 306 (678) 39,957	Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy Sell	2,990 6,617	30) (678) 39,957	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Sell Buy Sell	6,617	(678)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	389.848	39,957	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd IRS Securities Australia I td	lla/			
UBS Securities Australia Ltd UBS Securities Australia Ltd IRS Securities Australia I td	3	164,953	(16,929)	Ordinary
UBS Securities Australia Ltd	2	008 800	10,000	Ordinary.
I IRS Securities Australia 14d	200	131 650	ייייייייייייייייייייייייייייייייייייייי	Oldinary
	100	401,622	(0,00	Coldinary
IDC OLL LANGUAGE EXC	700	904,000	(175(17)	Ordinary
Ubs Global Asset Management	Buy	959,1	197	Ordinary
UBS Global Asset Management	Buy	89,200	8,062	Ordinary
UBS Securities Australia Ltd	Buy	341,250	35,000	Ordinary
UBS Securities Australia Ltd	Buy	31,483	3,229	Ordinary
UBS Securities Australia Ltd	Buy	23,390	2.399	Ordinary
UBS Securities Australia Ltd	Buy	49.949	5.123	Ordinary
UBS Securities Australia Ltd	Buv	192.767	19.771	Ordinary
UBS Securities Australia Ltd	Buv	16.731	1 716	Ordinary
LIBS Securities Australia I to	Bir	375 66	300.0	Ordinary
nat all the American Coo	Bay	015,22	5,235	Ordinary
UBS Securities Australia Ltd	Buy	19,500	2,000	Ordinary
UBS Securities Australia Ltd	Buy	1,004	103	Ordinary
UBS Securities Australia Ltd	8uy	5,158	529	Ordinary
UBS Securities Australia Ltd	Buy	19,003	1,953	Ordinary
UBS Securities Australia Ltd	Buy	7,259	746	Ordinary
UBS Securities Australia Ltd	Buy	428	44	Ordinary
UBS Securities Australia Ltd	Buv	76	101	Ordinary
UBS Securities Australia I td	Ruy	1 294	133	Ordinary
I IBS Securities Australia 110	Brit	430	27	Coling
I IRS Societies Australia I to	British	30 40	2000	Ordinaly
I IRS Societies Australia 1+d	600	C3r,12	4,402	Oldinary
ops seconties Australia Liu	bay	160	1,	Ordinary
UBS Securities Australia Ltd	Buy	20,540	2,111	Ordinary
UBS Securities Australia Ltd	Buy	78	8	Ordinary
UBS Securities Australia Ltd	Buy	3,026	311	Ordinary
UBS Securities Australia Ltd	Buy	847	87	Ordinary
UBS Securities Australia Ltd	Buv	2.851	793	Ordinary
LIRS Securities Australia 11d	Bin	26 534	707.0	Ording
Tipo Countries Australia Ltd	, ind	400,000	17/7	Orginaly
UBS Securities Australia Ltd	Buy	19,265	1,980	Ordinary
UBS Securities Australia Ltd	Buy	1,498	154	Ordinary
UBS Securities Australia ! td	8	1 1 2 9	116	Ordinan
ODS Securities Australia Ltd	áng	621,1	q.	Ordinary

9-Jun-14 U	UBS Securities Australia Ltd	٩	404	(41)	Ordinary
	IBS Securities Australia I to	3	404	(44)	i de di pato
ı	UBS Securities Australia Ltd	3	9,850	(1,000)	Ordinary
ı	BS Securities Australia Ltd		968	(16)	Ordinary
19-Jun-14 U	BS Securities Australia Ltd	lig.	996	(86)	Ordinary
П	UBS Securities Australia Ltd	Self	49	(2)	Ordinary
7	UBS Securities Australia Ltd	Sell	925'19	(6,245)	Ordinary
	UBS Securities Australia Ltd	Sell	1,785	(181)	Ordinary
┪	UBS Securities Australia Ltd	llas.	759	(77)	Ordinary
Ť	BS Securities Australia Ltd	llas.	2,349	(238)	Ordinary
۲	UBS Securities Australia Ltd	Jes Sell	3,168	(321)	Ordinary
9-Jun-14	BS Securities Australia Ltd	lia C	678	(84)	Ordinary
1	DS Securities Australia Ind	iiac o	9/8	(68)	Ordinary
	LIDE Cognition Australia 14d	Jac J	477'	(124)	Orginary
1	DS Securities Australia Ltd	nac J	4,10	(4 7)	Ordinary
Ť	DES SECURITIES AUSTRAINS LIQ	las.	1,/4/	(// 1)	Ordinary
19-July 14	UBS Securities Australia Ltd	- R	72,543	(7,337)	Ordinary
Ť	JDS Securities Australia Liu	ll Scill	35,493	(3,590)	Ordinary
1	RS Securities Australia (14)	la S	1312	(000,1)	Ordinary
	UBS Securities Australia Ltd	lleS	01	(6)	Ordinary
19-Jun-14	BS Securities Australia Ltd	las.	168	(21)	Ordinary
ı	UBS Securities Australia Ltd	lles.	10,925	(1,108)	Ordinary
19-Jun-14	UBS Securities Australia Ltd	Sell	16,930	(1,717)	Ordinary
	BS Securities Australia Ltd	ll-sc-ll	3,165	(321)	Ordinary
	UBS Securities Australia Ltd	}!es}	3)865	(365)	Ordinary
	BS Securities Australia Ltd	Seli	2,761	(280)	Ordinary
19-Jun-14 Ui	BS Securities Australia Ltd	Sell	4,240	(430)	Ordinary
-	UBS Securities Australia Ltd	Sell	4,467	(453)	Ordinary
	BS Securities Australia Ltd	Sell	10	(1)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	168	(17)	Ordinary
- 1	BS Securities Australia Ltd	Sell	3,116	(316)	Ordinary
19-Jun-14 UI	UBS Securities Australia Ltd	Sell	2,307	(234)	Ordinary
- 1	3S Securities Australia Ltd	Seli	414	(42)	Ordinary
7	JBS Securities Australia Ltd	Sell	3,165	(321)	Ordinary
Ī	JBS Securities Australia Ltd	Sell	2,600	(268)	Ordinary
7	UBS Securities Australia Ltd	Sell	2,307	(234)	Ordinary
9-Jun-14	JBS Securities Australia Ltd		1,322	(134)	Ordinary
- 1	3S Securities Australia Ltd	Sell	74,351	(2,533)	Ordinary
-1	UBS Securities Australia Ltd	Sell	11,390	(1,154)	Ordinary
-1	35 Securities Australia Ltd	Sell	18,990	(1,924)	Ordinary
	UBS Securities Australia Ltd	Seil	36,967	(3,753)	Ordinary
1	UBS Securities Australia Ltd	Sell	40,610	(4,127)	Ordinary
19-Jun-14 UE	UBS Securities Australia Ltd	Sell	18,499	(1,880)	Ordinary
7	3S Securities Australia Ltd	Sell	17,633	(1,792)	Ordinary
7	UBS Securities Australia Ltd	Sell	2,766	(286)	Ordinary
	UBS Securities Australia Ltd	Sell	354	(9E)	Ordinary
	UBS Securities Australia Ltd	Sell	7,823	(195)	Ordinary
ŀ	UBS Securities Australia Ltd	Sell	12,792	(1,300)	Ordinary
l	35 Securities Australia Ltd	Sell	8/8/9	(669)	Ordinary
1	3S Securities Australia Ltd	Sell	1,819	(185)	Ordinary
	35 Securities Australia Ltd	Sell	12,405	(1.262)	Ordinary
9-Jun-14 UE	UBS Securities Australia Ltd	Sell	265	(27)	Ordinary
9-Jun-14 UE	UBS Securities Australia Ltd	Sei	4.227	(430)	Ordinary
T	SS Securities Australia Ltd	9	4 591	(467)	Ordinary
9-Jun-14 UE	UBS Securities Australia Ltd	٥	4 129	(420)	Ordinary
Г	UBS Securities Australia Ltd	Sell	6.272	(829)	Ordinary
	UBS Securities Australia Ltd	les.	3,146	(320)	Ordinary
Ī	UBS Securities Australia Ltd	Sell	167	((1))	Ordinary
Ī	35 Securities Australia Itd	9	49	(5)	Ordinary
T	JBS Securities Australia 1td	Seil	0.835	(000 1)	Ordinary
Ť	IRS Societies Australia Ind	Call	CCO'S	(050)	Ordinary
T	S Societies Australia I+A	Cell	000,0	(000)	Ordinary
T	UBS Securities Australia Ltd	Sell	026'07	(7,222)	Ordinary
T	ODS Securities Australia Elia)ell	670,6	(0.10)	Cignary
		_	000 1	(E - L)	

1 m 1 7					-
3-Jun-14	UBS Securities Australia Ltd	Buy	7 391	750	Ordinary
Jun-14	UBS Securities Australia Ltd	Buv	18 359	1 883	Ordinary
3-km-14	LIBS Sequities Australia Ltd	Bry	13 338	1 260	Ordinary
3.1.0.17	TIDE Countition Australia Ltd	ÁDO G	19,330	905,1	Orginary
1 1 1	ing contines Australia Ltd	, and	10,304	Eqn'i	Ordinary
1 1 1	Obs securities Australia Lid	and	6///9	2005	Ordinary
- July - C	Obs securities Australia Ltd	, and	47,093	4,830	Ordinary
#1-Inc-C	UDS Securities Australia Ltd	any	77,974	2,864	Ordinary
1	OBS Securities Australia Ltd	Buy	5,850	009	Ordinary
5-Jun-14	UBS Securities Australia Ltd	Buy	2,291	235	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	43,700	4,482	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	806'08	3,170	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	3,900	400	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	52,094	5.343	Ordinary
3-lun-14	UBS Securities Australia Ltd	Bis	37.704	3000	
2 lim 1/	I IDC Cocurition August Ltd	, od	100.01	5,025	Clouday
1 2	Obs securities Australia Ltd	, and	19,295	6/6/1	Ordinary
3-JUN-14	UBS Securities Australia Ltd	Buy	3,900	400	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	70,239	7,204	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	6,825	2007	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	38,669	3,966	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	690'08	3.084	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	45.698	4.687	Ordinary
3-Jun-14	UBS Securities Australia Ltd	BIN	4 826	405	Ordinan
3-lin-14	185 Securities Australia 11d	(c)	000,5	2007	Corrain
3-lim-14	Libo Countries Australia Ltd	Suy P	006,6	400	Orginary
	CDC Carries Australia Lico	Say	20,212	3,714	Ordinary
41-15C-C	Obs securities Australia Ltd	Buy	7///07	2,131	Ordinary
100-14 100-14	UBS Securities Australia Ltd	Buy	19,929	2,044	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,492	153	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	3,900	400	Ordinary
3-Jun-14	UBS Securities Australia (1d	Bir	45 221	4 638	Ordinan
3-110-14	I IDS Societies Australia 14d	B	100 00	000,1	
	CDO Decurries Mostalia Ltd	edy.	160'07	2,/58	Ordinary
1-Inc-C	Opp Securities Australia Ltd	poo	1,706	1/2	Ordinary
4	UBS Securities Australia Ltd	Buy	25,769	2,643	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	19,744	2,025	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	22,854	2,344	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	8.235	845	Ordinary
un-14	UBS Securities Australia Ltd	\ne	370	38	Ordinary
3-hin-14	I IBS Sparities Australia 14d	(- B	2000	2,00	diam'n's
	CDS Securities Australia Liu	Áng	6,235	845	Ordinary
	ODS Securities Australia Ltd	hay	8,235	845	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	16,253	1,667	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	86	10	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,346	138	Ordinary
un-14	UBS Securities Australia Ltd	Bliv	2 945	302	Ordinan
3-lin-14	ILIBS Securities Australia Ltd	Buy	12 665	1 200	(in line)
	I DC Commission Accession and	, and	12,003	667'1	Granary
5-Jun-14	UBS Securities Australia Ltd	Buy	/31	75	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	6,767	694	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	30,225	3,100	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	4,339	445	Ordinary
un-14	UBS Securities Australia Etd	Buy	6 767	694	Ordinary
3-lin-14	(BS Sequities Australia (10)	Bir	30,025	0010	i i
	L4 - 1 - 1 - 4 1 - 1 - 1 - 1 - 1 - 1	65.7	2,000	201.	Clourary
11.00	Con Description Con	ána	006.6	400	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	26,910	2,760	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	497	51	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	3.764	386	Ordinary
11-14	I IRS Securities Australia 14d	Buy	27.5 3	100	or eight
	Late Contract Assistance 200	600	27010	- 600 -	Cidinary
2-101-0	Upo Securities Australia Lid	Buy	1,004	906,4	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,970	202	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	800	85	Ordinary
un-14	UBS Securities Australia Ltd	Buy	1,724	177	Ordinary
23-Jun-14	UBS Securities Australia Ltd	Suv	15,418	1.583	Ordinary
3-1 m-14	I IBS Securities Australia Ltd	6	1 675	524	
	The Contract Action Co.	, and	6/0,1	7/1	Courai
3-Jun-14	UBS Securities Australia Ltd	Buy	30,194	3,100	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,286	132	Ordinary
-Jun-14	UBS Securities Australia Ltd	ă	10100		
ļ		,000	30,134	3,100	Ordinary
23-Jun-14	UBS Securities Australia Ltd	Buv	30,134	3,100	Ordinary

Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Crainary
90	2,1/0	5,723	3,319	2	2	2	782	1,551	202	267	3,931	16	300	71	271	700	271	2 2	6 565	5,577	912	101	2,409	258	0,330	1.876	718	2,245	162	9,881	8 6	207	6	745	150	23	1,334	1845	400	3,586	268	270	267	5,872	3,702	6,232	1 015	277	188	282	2,864	280	266	796	907	28	495	3.249	0001
876	21,114	289,44	32,294	19	19	19	609'2	15,091	1,965	2,598	38,229	882	2,919	169	2,637	6,811	2,637	535	53.877	54,264	8,874	983	23,440	2,510	2 238	18.253	986'9	21,844	1,576	96,142	46/	2,032	29	7,249	1,460	224	12,980	1 790	3.892	34,856	2,605	2,624	2,595	57,252	36,095	4117	9,896	2,642	1.940	2,750	27,924	2,730	2,594	2,594	4,554	273	4.875	31.678	0 750
Buy	Ang P	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	buy.	818	Buy	Buy	Buy	Buy	Buy		Buy	Buy	Buy	Buy	Buy	Buy.	Buy	Buv	Buy	Bliv	Buv	Buy	Buy	Buy	8uy	Buy	Ana ana	B. B.	Buv	Buv	i i i													
UBS Securities Australia Ltd	IBC Countition Australia Ltd	DDS Securities Australia Ltd	DOS SECULIDES AUSUBIIA LICI	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia I td	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	ubs securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia (1d	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	JBS Securities Australia Itd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	I IRS Securities Australia I to
Ţ	23-Jun-14	T	1	23-Jun-14	7	7	- 1	- 1	- 1	23-Jun-14 (Ť		7	23-Jun-14	T	T	23-Jun-14	T	Ť	23-Jun-14	-		7	23-Jun-14	ł		23-Jun-14	Ħ	T	Ť	23-Jun-14	Ť	Ī	Ī	Ī	1	23-Jun-14 L	T	1		H	1	٦	- 1	23-Jun-14 U	1	ì	T	T	Ť	23-Jun-14 U	П	7	23-Jun-14 Ju	T	T	Ī	T	Т

23-Jun-14 23-Jun-14 23-Jun-14	UBS Securities Australia Ltd	Buy	273	28	Ordinary
23-Jun-14 23-Jun-14	I IRS Securities Australia Ltd				
3-Jun-14	מס מכתונות ויתחוות דות	Buy	1,375	141	Ordinary
	UBS Securities Australia Ltd	Buy	137	14	Ordinary
23-Jun-14	UBS Securities Australia Ltd	Buy	2,213	227	Ordinary
23-Jun-14	UBS Securities Australia Ltd	Biv	7256	1 023	Ordinary
3-lin-14	I IBS Securities Australia I to	Bary	7116	710	Ciginal y
3-Jun-14	LIBS Securities Australia Ltd	Ban.	13 102	1 353	Crainary
3-hin-14	LIBS Sequitities Australia Ltd	Briv	3000	000	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Biv	1 873	791	Ordinary
3-Jun-14	LIBS Securities Australia Ltd	Birv	EV8 8	200	Ordinary
3-lun-14	LIBS Securities Australia 11d	Bire	2,072	210	Colonia
23-lim-14	LIRS Securities Australia 11d	in a	7,0,7	213	Ordinary
3-110-14	I IDC Cocurition Australia and	600	41 C/C	7	Cidinary
3-lin-14	I IRC Societies Australia 144	600	000 6	01/	Orginary
23-1 m-14	I IDS Countition Australia Ltd	ony 9.00	006,5	3	Orginary
1 1 1 1 1	Libra Securities Australia Ltd	Sna S	D	243	Orginary
1	UBS Securities Australia Ltd	Buy	11,244	570,21	Ordinary
-	UBS Securities Australia Ltd	Buy	614	63	Ordinary
1	UBS Securities Austrana Ltd	Buy	796'9	/14	Ordinary
	Obs securities Australia Ltd	Buy	666,1	502	Ordinary
1	Obs securities Australia Ltd	Buy	014	42	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	2,564	263	Ordinary
-	UBS Securities Australia Ltd	Buy	2,555	797	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,326	136	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,316	135	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,346	138	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,336	137	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,336	137	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,336	137	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,365	140	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,365	140	Ordinary
1	UBS Securities Australia Ltd	Buv	410	42	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	1 376	136	Ordinary
ı	UBS Securities Australia Ltd	Buv	1355	139	Ordinary
1	II IRS Securities Australia i to	Brit	3751	130	Commen
23-Jun-14	UBS Securities Australia 1td	B _i	75.6	137	Ordinary
3-lin-14	LIRS Securities Australia Ltd	Bux	80	5	Ordinary
12 him 14	1 IDC Constitute Australia 1+a		000	2	oldinary Callinary
1 1 1 1 C	The Court of the State of the S	, and	925,1	951	Orginary
1 1 1	Obs seculities Australia Liu	and a	925,1	30	Ordinary
5-Juli-14	ODS Securities Australia Ltd	euv.	1,336	13/	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	176	18	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,101	113	Ordinary
	UBS Securities Australia Ltd	Buy	51,892	5,325	Ordinary
	UBS Securities Australia Ltd	Buy	1,715	176	Ordinary
l	UBS Securities Australia Ltd	Buy	9.764	1,002	Ordinary
	UBS Securities Australia Ltd	Buy	25 951	2,663	Ordinary
	I IBS Societities Australia I +d	0	010	700	(in the
10 1 C	I IDC Committee A control and	ing.	510	4 6	Crainary
-1-Inc-c	Obs securities Australia Ltd	pay	3/3/5/2	38,313	Ordinary
1	UBS Securities Australia Ltd	BUy	6,4/4	900	Ordinary
١	UBS Securities Australia Ltd	Buy	1,316	135	Ordinary
1	UBS Securities Australia Ltd	Buy	5,772	265	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	5,372	551	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	13,962	1,432	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	3.900	400	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	205	12	Ordinary
ı	UBS Securities Australia 11d	Bliv	103 828	10.649	Ordinary
ı	UBS Securities Australia Ltd	Birk	3715	381	Cremina
ı	UBS Securities Australia I td	Buy	5 372	551	Ordinary
1	UBS Securities Australia Ltd	Buy	727	23	Ordinary
ı	LIBS Sporitips Australia Ltd	R. I	4 573	7460	Cochio
	I IRC Constitute Australia 14d	Berry	2,632	020	Ordinary
2-July 14	ODS SECULIES AUSTRAINA LIC	éng.	2,033	0/7	Ordinary
1	UBS Securities Australia Ltd	Buy	614	63	Ordinary
-	UBS Securities Australia Ltd	Buy	1,638	168	Ordinary
-	UBS Securities Australia Ltd	Buy	4,466	458	Ordinary
	UBS Securities Australia Ltd	Buy	887	10	Ordinan
23-1 in 14			}	1	, E 1815
1	UBS Securities Australia Ltd	Buy	6 143	630	Ordinary

1,997 1,997 1,997 1,997 1,997 1,997 1,1997 1,1997 1,997 1,997 1,997	672 693 672 69 672 69 3,204 309 3,120 379 3,127 321 3,127 321 3,127 321 3,127 321 3,127 321 3,127 321 3,127 321 3,284 409 3,284 409 3,068 246 2,448 409 3,068 315 3,068 315 3,068 315 3,07 315 2,44 25 2,44 25 3,068 315 3,068 315 2,544 25 2,526 74 2,536 790 7,759 790 7,759 790 7,759 790 7,190 225 2,445 2,67 3,896 <t< th=""></t<>
1997 1997 1978 1978 1978 1789	
3,204 19,480 1,127 3,127 3,127 3,127 3,127 3,127 3,688 1,140 2,396 2,396 3,068 3,068 3,068 1,759 1,140 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,517 1,032 1,759	
19,480 6 68 1,128 1,1286 1,1286 1,286 1,286 1,286 2,396 2,396 2,416	
3,127 3,419 5,698 1,140 1,140 1,286 3,984 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,416 2,517 1,032 1,032 1,573 1,595 1,573 1,595 1,573 1,595 1,573 1,595 1,573 1,595 1,573 1,595 1,573 1,595 1,596 2,690	
3,127 3,127 5,598 1,286 1,286 1,286 1,286 2,396 2,396 1,140 1,140 1,140 1,140 2,290 2,390 2,390 2,396 2,390 2,390 2,390 1,140 1,	
3,419 5,698 1,140 1,1286 3,984 3,984 2,396 2,396 3,068 3,068 3,068 2,396 1,140 2,396 2,396 2,396 2,396 2,396 2,396 2,396 2,396 2,396 1,140 2,290 2	
5,698 1,140 1,140 1,140 2,416 2,416 3,068	
769 1,140 1,286 3,984 2,396 2,396 3,068 3,	
1,140 1,	
1,286 3,984 3,984 2,446 2,446 3,068 3,068 3,068 3,044 2,44 3,33 1,140 8,435 3,068 3,	
3,984 3,984 3,068 3,	
2,149 2,149 2,146 2,416 3,068	
2,396 2,396 2,396 3,068 3,068 3,140 1,140 1,140 2,390 2,390 2,390 2,571 1,037 1,595 1,575 1,575 1,575 1,575 1,759	
2,796 448 2,416 2,616 2,068 4,393 4,140 8,435 8,435 8,435 8,435 8,435 1,590 1,255 1,256 1,255 1,25	
2,416 2,416 3,626 3,626 3,636 1,140 8,433 3,024 2,930 7,256 6,990 1,591 1,032 1,575	
2.416 2.68 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068	
2,416 2,26 3,068 4,393 4,393 8,144 2,844 2,844 2,940 7,290 6,590 6,590 1,575 1,575 1,575 1,575 1,575 1,595 1,5	
526 3,068 4,393 1,140 8,435 8,435 902 2,290 2,290 2,290 2,290 2,290 2,290 2,290 2,290 1,097 1,097	
3,068 4,393 1,140 8,4,140 8,4,140 1,2,430 1,2,930 1,7,759 1,759 1,759 1,759 1,759 1,759 1,759 1,759 1,	
3,088 4,393 1,140 8,435 30.2 5,844 2,930 2,930 6,990 1,759 1	
1,140 8,435 302 302 304 2,44 2,44 2,930 6,930 1,525 1,525 1,525 1,525 1,525 1,525 1,535 1,997	
1,140 8,435 302 5,844 2,930 7,256 6,990 5,571 100,322 1,759 1,	
8,435 302 304 5,844 2,444 2,430 2,530 6,930 1,557 1,525 1,525 1,525 1,525 1,525 1,525 1,535 1,68 3,068 3,068 3,068 1,997	
302 2,44 2,430 7,256 6,990 6,990 1,759 1,575 1,575 1,575 1,575 3,068 3,068 3,068 3,068 3,068 3,068 3,068 1,995 1,997	
5.844 244 244 2.230 7.256 6.990 6.990 7.759 1.759 1.759 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068 3.068	
5,844 2,940 7,256 6,990 5,571 100,322 1,575 1,525 3,068 3,068 3,068 3,068 1,997 1,997	
244 2.250 7.256 6.930 6.930 7.759 15.255 3.068 3.068 6.993 1,997	
2,930 6,930 6,932 10,759 15,255 1,759 3,068 3,088 6,993 1,997	
7,256 6,990 100,322 7,759 7,759 1,086 9,068 9,068 9,068 1,997	
5,571 (5,930) (0,322) (0,322) (1,525) (1,525) (1,525) (1,680) (1,997)	
6,990 6,990 100,322 7,759 11,225 3,068 3,896 6,993 1,997	
5,571 100,322 1759 15,255 3,068 3,896 6,993 1,997	
100,322 7,759 15,255 3,068 3,896 6,993 1,997	
7,759 15,255 3,068 3,896 6,993 1,997	
15,735 3,068 3,896 6,993 1,997	
15,255 3,068 3,896 6,993 1,997	555 906 907 907 907 907 907 907 908 909 909 909 909 909 909 909
3,068 3,896 6,993 1,997	988 996 990 990 995 995 995 997 997 996 996
3,896 6,993 1,997	996 990 990 975 975 975 975 976
6,993	33 35 55 76 76 76
1,997	37 35 35 78 76 87 87
100	25 2 3 3 5 7 7 8 8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9
	25 5 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5
2,190	35 54 76 57 36
2,435	54 76 57 37
4,354	78 76 37 36
8678	76 57 37 36
377.0	37 38
50, 5)7 36
6,467	36
209'2	36
96,536	
19 5	5
000	
1,490	
12,204	1,253
584	
400	
COT	
166,2	
1,179	20
14 513	
0100	
062,2	
4,441	
1 227	126
1337	
1,383	53
6,516	9
3 89K	32
0707	
4,149	
26,949	5,764
2,584	
623	0
000,	
31,088	05,25U
1,463	
11,700	
-	201
	2.591 1,2,204 2,584 4,00 1,1,79 1,1,79 1,4,51 1,27 1,27 1,32

USS Scontines Australia LID Buy 799 R2 USB Scontines Australia LID Buy 799 R2 USB Scontines Australia LID Buy 769 79 79 USB Scontines Australia LID Buy 7,762 79 USB Scontines Australia LID Buy 7,69 77 USB Scontines Australia LID Buy 7,69 77 USB Scontines Australia LID Buy 7,69 77 USB Scontines Australia LID Buy 7,69 7,71 USB Scontines Australia LID Buy 7,69 7,71 USB Scontines Australia LID Buy 7,70 7,20 USB Scontines Australia LID Buy 7,70 7,20 USB Scontine Aus	3-Jun-14	LIDE Contribute Australia 144	600	100	. 6	Oldinary
Ligs Securities Australia Lid Buy 769 779 Ligs Securities Australia Lid Buy 769 779 Ligs Securities Australia Lid Buy 769 779 Ligs Securities Australia Lid Buy 776 779 Ligs Securities Australia Lid Buy 7,769 779 Ligs Securities Australia Lid Buy 2,075 721 Ligs Securities Australia Lid Buy 2,075 720 Ligs Securities Australia Lid Buy 1,275 1,28 Ligs Securities Australia Lid Buy 1,275 1,271 Ligs Securities Australia Lid Buy 1,275 1,271 Ligs Securities Australia Lid Buy	-lin-14	IIRS Securities Australia I to	200	902	700	Ordinary
USS Securities Australia Ltd Buy 7789 770 77	-Jun-14	UBS Securities Australia Ltd	Buy	692	20 02	Ordinary
UBS Securites Australia Ltd BM 7.95 7.9 UBS Securites Australia Ltd BM 7.62 7.9 UBS Securites Australia Ltd BM 7.62 7.9 UBS Securites Australia Ltd BM 7.2 7.9 UBS Securites Australia Ltd BM 2.211 2.27 UBS Securites Australia Ltd BM 2.211 2.27 UBS Securites Australia Ltd BM 2.211 2.27 UBS Securities Australia Ltd BM 2.211 2.27 UBS Securities Australia Ltd BM 2.211 2.27 2.8 UBS Securities Australia Ltd BM 2.201 3.9 3.9 3.9 3.9 UBS Securities Australia Ltd BM 2.202 3.2 3.0 3.2 3.0 3.2 3.0 3.2 3.0 3.2 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0 3.0	-Jun-14	UBS Securities Australia I td	Buy	087	5 6	Ordinary
(45) Securities Australia Ltd Buy 769 779 (45) Securities Australia Ltd Buy 769 779 (45) Securities Australia Ltd Buy 1,782 773 (46) Securities Australia Ltd Buy 2,075 273 (47) Securities Australia Ltd Buy 2,075 273 (48) Securities Australia Ltd Buy 2,075 273 (48) Securities Australia Ltd Buy 2,075 273 (48) Securities Australia Ltd Buy 2,075 273 (48) Securities Australia Ltd Buy 2,075 2,775 (48) Securities Australia Ltd Buy 2,075 2,775 (48) Securities Australia Ltd Buy 1,246 2,777 (48) Securities Australia Ltd Buy 1,246 2,777 (48) Securities Australia Ltd Buy 1,246 2,777 (48) Securities Australia Ltd Buy 4,136 1,247 (48) Securities Australia Ltd Buy 1,246 2,777 (48) Securities Australia Ltd B	1	1 los constitut Augustia Lau	kng	60/	io i	Ordinary
89 99 769 789 80 30 Securities Australia Ltd Buy 3,438 789 80 30 Securities Australia Ltd Buy 2,207 213 80 20 Securities Australia Ltd Buy 2,201 227 10 20 Securities Australia Ltd Buy 2,076 613 10 20 Securities Australia Ltd Buy 2,076 2,077 10 20 Securities Australia Ltd Buy 2,076 613 10 20 Securities Australia Ltd Buy 2,076 1,076 10 30 Securities Australia Ltd Buy 1,076 1,077 10 30 Securities Australia Ltd Buy 1,078 1,077 10 30 Securities Australia Ltd Buy 1,078 1,077	1 1 1	Obs Securities Australia Lita	Buy	69/	5	Ordinary
QS Securities Australia Ltd Buy 1,723 1,33 1,33 QS Securities Australia Ltd Buy 2,075 2,13 LBS Securities Australia Ltd Buy 2,075 2,13 UBS Securities Australia Ltd Buy 2,09 79 LBS Securities Australia Ltd Buy 2,69 77 LBS Securities Australia Ltd Buy 2,63 1,54 LBS Securities Australia Ltd Buy 2,63 1,54 LBS Securities Australia Ltd Buy 1,246 1,24 LBS Securities Australia Ltd Buy 1,246 1,24 LBS Securities Australia Ltd Buy 1,246 1,24 LBS Securities Australia Ltd Buy 1,246 1,25 LBS Securities Australia Ltd Buy 1,246 1,25 LBS Securities Australia Ltd Buy 1,26 2,77 LBS Securities Australia Ltd Buy 1,26 2,77 LBS Securities Australia Ltd Buy 1,156 2,27 LBS Securities Australia Ltd Bu	1 1 1	Obs Securities Australia Lid	ng	69/	6/	Ordinary
UBS Securities Australia Ltd Buy 2,075 183 UBS Securities Australia Ltd Buy 2,075 2.17 2.27 UBS Securities Australia Ltd Buy 2,075 2.27 2.27 UBS Securities Australia Ltd Buy 7,09 1.54 2.27 2.	100	Obs Securities Australia LIG	Buy	3,438	353	Ordinary
UBS Securities Australia Ltd Buy 2,171 2,27 UBS Securities Australia Ltd Buy 2,171 2,17 2,17 UBS Securities Australia Ltd Buy 2,149 7,29 7,29 UBS Securities Australia Ltd Buy 2,149 4,25 2,28 UBS Securities Australia Ltd Buy 2,149 4,25 1,29 UBS Securities Australia Ltd Buy 2,171 1,175 1,175 1,176 UBS Securities Australia Ltd Buy 2,174 2,127 1,275 1,176	101-14	UBS Securities Australia Ltd	Buy	1,782	183	Ordinary
Qual Securities Australia Ltd Buy 2,271 2,27 1 (BS) Securities Australia Ltd Buy 7,968 61.3 1 (BS) Securities Australia Ltd Buy 1,279 42.5 1 (BS) Securities Australia Ltd Buy 1,279 42.5 1 (BS) Securities Australia Ltd Buy 4,140 1,717 1 (BS) Securities Australia Ltd Buy 4,140 1,717 1 (BS) Securities Australia Ltd Buy 4,140 1,246 1 (BS) Securities Australia Ltd Buy 4,120 1,28 1 (BS) Securities Australia Ltd Buy 4,120 1,28 1 (BS) Securities Australia Ltd Buy 8,00 3,00 3,20 1 (BS) Securities Australia Ltd Buy 4,126 1,20	1 - 100	Obs Securities Australia Ltd	Buy	2,0/5	213	Ordinary
UBS Securities Australia Ltd Buy 5,566 6,13 UBS Securities Australia Ltd Buy 1,499 1,499 1,59 UBS Securities Australia Ltd Buy 2,140 4,25 1,69 UBS Securities Australia Ltd Buy 2,140 4,25 1,27 UBS Securities Australia Ltd Buy 1,175 1,75 1,75 UBS Securities Australia Ltd Buy 1,274 1,28 UBS Securities Australia Ltd Buy 1,275 1,75 UBS Securities Australia Ltd Buy 1,276 1,75 UBS Securities Australia Ltd Buy 1,276 1,75 UBS Securities Australia Ltd Buy 2,070 3,26 UBS Securities Australia Ltd Buy 3,090 3,270 UBS Securities Australia Ltd Buy 4,120 4,220 UBS Securities Australia Ltd Buy 1,126 1,27 UBS Securities Australia Ltd Buy 4,286 3,27 UBS Securities Australia Ltd Buy 1,156 1,27 <td>14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>2,211</td> <td>227</td> <td>Ordinary</td>	14	UBS Securities Australia Ltd	Buy	2,211	227	Ordinary
UBS Securities Australia Ltd Buy 5,968 61 3 UBS Securities Australia Ltd Buy 1,299 1,51 UBS Securities Australia Ltd Buy 2,635 2,77 UBS Securities Australia Ltd Buy 2,645 2,77 UBS Securities Australia Ltd Buy 2,645 1,246 1,78 UBS Securities Australia Ltd Buy 2,645 1,25 1,28 UBS Securities Australia Ltd Buy 2,047 3,297 3,38 UBS Securities Australia Ltd Buy 8,04 1,266 1,28 UBS Securities Australia Ltd Buy 8,05 3,297 3,38 UBS Securities Australia Ltd Buy 8,05 1,126 1,26 1,27 UBS Securities Australia Ltd Buy 8,05 1,156 1,27 1,24 UBS Securities Australia Ltd Buy 1,156 1,23 1,24 UBS Securities Australia Ltd Buy 1,156 1,23 UBS Securities Australia Ltd Buy 1,156 1,23 <td>4</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>692</td> <td>79</td> <td>Ordinary</td>	4	UBS Securities Australia Ltd	Buy	692	79	Ordinary
UBS Securities Australia Ltd Buy 1,499 154 UBS Securities Australia Ltd Buy 2,171 28 UBS Securities Australia Ltd Buy 2,171 175 UBS Securities Australia Ltd Buy 2,171 176 UBS Securities Australia Ltd Buy 2,071 1,071 UBS Securities Australia Ltd Buy 2,071 1,071 UBS Securities Australia Ltd Buy 2,072 3,289 UBS Securities Australia Ltd Buy 1,072 1,075 UBS Securities Australia Ltd Buy 4,136 4,229 UBS Securities Australia Ltd Buy 4,136 4,239 UBS Securities Australia Ltd Buy 4,136 4,239 UBS Securities Australia Ltd Buy 4,136 1,26 1,27 UBS Securities Australia Ltd Buy 4,136 4,239 1,27 UBS Securities Australia Ltd Buy 3,206 3,27 1,20 UBS Securities Australia Ltd Buy 1,166 1,20 1,20 <td>-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>2,968</td> <td>613</td> <td>Ordinary</td>	-Jun-14	UBS Securities Australia Ltd	Buy	2,968	613	Ordinary
UBS Securities Australia Ltd Buy 4,173 2,8 UBS Securities Australia Ltd Buy 4,174 4,25 UBS Securities Australia Ltd Buy 4,172 1,174 1,178 UBS Securities Australia Ltd Buy 2,64 2,171 1,28 UBS Securities Australia Ltd Buy 2,64 3,290 8,29 UBS Securities Australia Ltd Buy 8,00 8,29 3,315 1,28 UBS Securities Australia Ltd Buy 8,00 8,29 3,315 1,28 UBS Securities Australia Ltd Buy 4,128 4,229 4,229 UBS Securities Australia Ltd Buy 4,138 4,226 1,271 UBS Securities Austra	-Jun-14	UBS Securities Australia Ltd	Buy	1,499	154	Ordinary
UBS Securities Australia Ltd Buy 4,140 4,15 UBS Securities Australia Ltd Buy 1,171 1,72 1,71 1,72 1,71 1,72 1,72 1,71 1,72 1,72 1,72 1,72 1,71 1,72 1,72 1,72 1,72 1,72 1,72 <t< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>273</td><td>28</td><td>Ordinary</td></t<>	-Jun-14	UBS Securities Australia Ltd	Buy	273	28	Ordinary
UBS Securities Australia Ltd Bluy 1,919 1,919 UBS Securities Australia Ltd Bluy 26,436 2,171 UBS Securities Australia Ltd Bluy 1,216 1,715 UBS Securities Australia Ltd Bluy 1,216 1,726 UBS Securities Australia Ltd Bluy 8,000 8,229 UBS Securities Australia Ltd Bluy 8,000 8,200 8,200 UBS Securities Australia Ltd Bluy 1,156 1,201 1,201 UBS Securities Australia Ltd <th< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>4,140</td><td>425</td><td>Ordinary</td></th<>	-Jun-14	UBS Securities Australia Ltd	Buy	4,140	425	Ordinary
UBS Securities Australia Ltd Buy 26.436 2.717 UBS Securities Australia Ltd Buy 1.246 1.246 1.246 1.246 1.246 1.245 1.277 1.235 1.247 1.235 1.247 1.235 1.246 1.245 1.247 1.245 1.247 1.248 2.248<	-Jun-14	UBS Securities Australia Ltd	Buy	1,919	197	Ordinary
UBS Securities Australia Ltd Buy 1,712 178 UBS Securities Australia Ltd Buy 1,722 178 UBS Securities Australia Ltd Buy 1,247 1,28 UBS Securities Australia Ltd Buy 3,290 8,280 UBS Securities Australia Ltd Buy 8,020 8,281 UBS Securities Australia Ltd Buy 8,020 8,281 UBS Securities Australia Ltd Buy 4,320 8,281 UBS Securities Australia Ltd Buy 4,186 4,322 UBS Securities Australia Ltd Buy 8,02 3,28 UBS Securities Australia Ltd Buy 4,186 4,320 UBS Securities Australia Ltd Buy 1,166 1,27 UBS Securities Australia Ltd Buy 1,166 1,247 UBS Securities Australia Ltd Buy </td <td>-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>26,436</td> <td>2,717</td> <td>Ordinary</td>	-Jun-14	UBS Securities Australia Ltd	Buy	26,436	2,717	Ordinary
UBS Securities Australia Ltd Bluy 1,246 128 UBS Securities Australia Ltd Buy 1,247 1,25 UBS Securities Australia Ltd Buy 32,907 3,389 UBS Securities Australia Ltd Buy 4,320 3,287 UBS Securities Australia Ltd Buy 3,500 3,287 UBS Securities Australia Ltd Buy 4,136 4,213 UBS Securities Australia Ltd Buy 3,500 3,500 UBS Securities Australia Ltd Buy 3,500 3,500 UBS Securities Australia Ltd Buy 3,500 3,201 UBS Securities Australia Ltd Buy 3,600 3,201 UBS Securities Australia Ltd Buy 3,600 3,201 UBS Securities Australia Ltd Buy 1,156 1,27 UBS Securities Australia Ltd <th< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1,712</td><td>176</td><td>Ordinary</td></th<>	-Jun-14	UBS Securities Australia Ltd	Buy	1,712	176	Ordinary
UBS Securities Australia Ltd Buy 1,217 125 UBS Securities Australia Ltd Buy 32,907 3,898 UBS Securities Australia Ltd Buy 4,202 3,829 UBS Securities Australia Ltd Buy 4,120 3,810 3,812 UBS Securities Australia Ltd Buy 4,120 3,810 3,81 UBS Securities Australia Ltd Buy 4,120 4,229 UBS Securities Australia Ltd Buy 4,120 4,220 UBS Securities Australia Ltd Buy 4,128 4,266 UBS Securities Australia Ltd Buy 4,128 4,266 UBS Securities Australia Ltd Buy 4,128 4,266 UBS Securities Australia Ltd Buy 1,156 1,23 UBS	-Jun-14	UBS Securities Australia Ltd	Buy	1,246	128	Ordinary
UBS Securities Australia Ltd Buy 32,907 3,389 UBS Securities Australia Ltd Buy 8,000 8,25 UBS Securities Australia Ltd Buy 4,1,905 4,329 UBS Securities Australia Ltd Buy 4,1,905 4,329 UBS Securities Australia Ltd Buy 3,260 3,315 UBS Securities Australia Ltd Buy 4,1,305 3,280 4,329 UBS Securities Australia Ltd Buy 4,1,305 3,280 4,213 UBS Securities Australia Ltd Buy 4,1,305 3,280 4,213 UBS Securities Australia Ltd Buy 4,1,766 1,20 UBS Securities Australia Ltd Buy 3,200 3,328 4,213 UBS Securities Australia Ltd Buy 4,180 1,176 1,20 UBS Securities Australia Ltd Buy 2,479 1,24 UBS Securities Australia Ltd Buy 2,479 1,26 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd <td>-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>1,217</td> <td>125</td> <td>Ordinary</td>	-Jun-14	UBS Securities Australia Ltd	Buy	1,217	125	Ordinary
UBS Securities Australia Ltd Buy 41,881 42,23 UBS Securities Australia Ltd Buy 41,881 4,329 UBS Securities Australia Ltd Buy 41,380 4,329 UBS Securities Australia Ltd Buy 38,105 4,329 UBS Securities Australia Ltd Buy 38,10 39,4 UBS Securities Australia Ltd Buy 41,380 4,266 UBS Securities Australia Ltd Buy 1,166 1,20 UBS Securities Australia Ltd Buy 1,166 1,20 UBS Securities Australia Ltd Buy 1,166 1,20 UBS Securities Australia Ltd Buy 1,166 1,27 UBS Securities Australia Ltd Buy 1,166 1,27 UBS Securities Australia Ltd Buy 1,186 1,27 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Buy	32,907	3,389	Ordinary
UBS Securities Australia Ltd Buy 41 861 4.329 UBS Securities Australia Ltd Buy 4,1305 3.415 UBS Securities Australia Ltd Buy 3,105 3.94 UBS Securities Australia Ltd Buy 4,1305 4,213 UBS Securities Australia Ltd Buy 4,0782 4,213 UBS Securities Australia Ltd Buy 4,0782 4,213 UBS Securities Australia Ltd Buy 4,1380 4,266 UBS Securities Australia Ltd Buy 4,1380 4,213 UBS Securities Australia Ltd Buy 1,166 1,20 UBS Securities Australia Ltd Buy 1,166 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd <t< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>8,020</td><td>826</td><td>Ordinary</td></t<>	-Jun-14	UBS Securities Australia Ltd	Buy	8,020	826	Ordinary
UBS Securities Australia Ltd Buy 41.905 4.329 UBS Securities Australia Ltd Buy 3.066 3.315 UBS Securities Australia Ltd Buy 3.610 3.516 UBS Securities Australia Ltd Buy 40.782 4.213 UBS Securities Australia Ltd Buy 40.782 4.266 UBS Securities Australia Ltd Buy 40.782 4.213 UBS Securities Australia Ltd Buy 40.782 4.213 UBS Securities Australia Ltd Buy 1.166 1.20 UBS Securities Australia Ltd Buy 1.166 1.20 UBS Securities Australia Ltd Buy 1.166 1.247 UBS Securities Australia Ltd Buy 1.166 1.23 UBS Securities Australia Ltd Buy 1.166 1.247 UBS Securities Australia Ltd Buy 1.156 1.247 UBS Securities Australia Ltd Buy 1.156 1.25 UBS Securities Australia Ltd Buy 1.156 1.25 UBS Securities Australia Ltd <	-Jun-14	UBS Securities Australia Ltd	Buy	41,861	4,329	Ordinary
UBS Securities Australia Ltd Buy 3,056 3,315 UBS Securities Australia Ltd Buy 3,610 3,94 UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 41,380 4,266 UBS Securities Australia Ltd Buy 41,380 4,266 UBS Securities Australia Ltd Buy 1,166 1,20 UBS Securities Australia Ltd Buy 1,166 1,23 UBS Securities Australia Ltd Buy 1,176 1,23 UBS Securities Australia Ltd Buy 2,4,98 2,4,78 UBS Securities Australia Ltd Buy 1,176 1,23 UBS Securities Australia Ltd Buy 1,176 1,23 UBS Securities Australia Ltd Buy 1,176 1,23 UBS Securities Australia Ltd Buy 1,176 1,27 UBS Securities Australia Ltd <t< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>41,905</td><td>4,329</td><td>Ordinary</td></t<>	-Jun-14	UBS Securities Australia Ltd	Buy	41,905	4,329	Ordinary
UBS Securities Australia Ltd Buy 3,810 394 UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 1,27 UBS Securities Australia Ltd Buy 1,196 1,247 UBS Securities Australia Ltd Buy 1,196 1,27 UBS Securities Australia Ltd Buy 1,196 1,24 UBS Securities Australia Ltd Buy 1,196 1,24 UBS Securities Australia Ltd Buy 1,196 1,24 UBS Securities Australia Ltd Buy 4,180 1,26 UBS Securities Australia Ltd Buy	-jun-14	UBS Securities Australia Ltd	Buy	32,056	3,315	Ordinary
UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 40,782 4,286 UBS Securities Australia Ltd Buy 40,782 4,286 UBS Securities Australia Ltd Buy 1,566 1,20 UBS Securities Australia Ltd Buy 1,566 1,00 UBS Securities Australia Ltd Buy 1,566 1,00 UBS Securities Australia Ltd Buy 1,566 1,00 UBS Securities Australia Ltd Buy 1,605 2,017 UBS Securities Australia Ltd Buy 1,366 1,23 UBS Securities Australia Ltd Buy 1,366 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,24 UBS Securities Australia Ltd Buy	-Jun-14	UBS Securities Australia Ltd	Buy	3,810	394	Ordinary
UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 41,380 4,213 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,267 107 UBS Securities Australia Ltd Buy 1,266 1,20 UBS Securities Australia Ltd Buy 1,265 1,247 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,256 1,24 UBS Securities Australia Ltd Buy 1,255 1,26 UBS Securities Australia Ltd Buy 1,255 1,27 UBS Securities Australia Ltd Buy	-Jun-14	UBS Securities Australia Ltd	Buv	145	15	Ordinary
UBS Securities Australia Ltd Buy 40,782 4,213 UBS Securities Australia Ltd Buy 4,266 120 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,176 123 UBS Securities Australia Ltd Buy 4,180 1,27 UBS Securities Australia Ltd Buy 4,180 1,27 UBS Securities Australia Ltd Buy 1,266 127 UBS Securities Australia Ltd Buy 1,266 127 UBS Securities Australia Ltd Buy 1	-Jun-14	UBS Securities Australia Ltd	Buy	803	c x	Ordinary
UBS Securities Australia Ltd Buy 41,380 4,266 UBS Securities Australia Ltd Buy 1,166 100 UBS Securities Australia Ltd Buy 1,166 100 UBS Securities Australia Ltd Buy 1,205 393 UBS Securities Australia Ltd Buy 1,366 1,07 UBS Securities Australia Ltd Buy 1,196 1,03 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,24 UBS Securities Australia Ltd Buy 1,196 1,24 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy	-Jun-14	UBS Securities Australia Ltd	Buv	40.782	4 213	Ordinary
UBS Securities Australia Ltd Buy 3.528 3.528 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,267 1,247 UBS Securities Australia Ltd Buy 1,176 1,247 UBS Securities Australia Ltd Buy 1,176 1,23 UBS Securities Australia Ltd Buy 24,096 2,499 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,206 1,20 UBS Securities Australia Ltd Buy	-Jun-14	UBS Securities Australia Ltd	Buv	41 380	4 266	Ordinary
UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 1,247 UBS Securities Australia Ltd Buy 1,1960 2,473 UBS Securities Australia Ltd Buy 1,196 1,247 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 2,4,096 2,479 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 4,180 1,24 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,260 1,07 UBS Securities Australia Ltd Buy 1,253,800 1,27,00 UBS Securities Australia Ltd Buy 1,260 1,07 UBS Securities Australia Ltd B	-Jun-14	UBS Securities Australia Ltd	Buy	825'	363	Ordinan
UBS Securities Australia Ltd Buy 1,70 1,70 UBS Securities Australia Ltd Buy 1,166 120 UBS Securities Australia Ltd Buy 1,166 1,20 UBS Securities Australia Ltd Buy 1,136 1,247 UBS Securities Australia Ltd Buy 1,136 1,23 UBS Securities Australia Ltd Buy 1,176 1,13 UBS Securities Australia Ltd Buy 1,176 1,1 UBS Securities Australia Ltd Buy 1,25,8 840 UBS Securities Australia Ltd Buy 1,25 1,27 UBS Securities Australia Ltd Buy 1,25,8 1,27 UBS Securities Australia Ltd Buy 1,25,8 1,27 UBS Securities Australia Ltd Sell	14 m-14	I IRS Securities Australia I to	620	1355	500	Coming
UBS Securities Australia Ltd Buy 1.56 1.6 UBS Securities Australia Ltd Buy 1.166 120 UBS Securities Australia Ltd Buy 1.21 1.247 UBS Securities Australia Ltd Buy 1.21 1.247 UBS Securities Australia Ltd Buy 6.075 2.4.99 UBS Securities Australia Ltd Buy 2.4.096 2.4.93 UBS Securities Australia Ltd Buy 4.180 1.23 UBS Securities Australia Ltd Buy 4.180 1.23 UBS Securities Australia Ltd Buy 1.196 1.23 UBS Securities Australia Ltd Buy 1.196 1.23 UBS Securities Australia Ltd Buy 1.255.889 1.29 000 UBS Securities Australia Ltd Buy 1.255.889 1.20 000 UBS Securities Australia Ltd Buy 1.255.889 1.20 000 UBS Securities Australia Ltd Buy 8.266 8.27 00 UBS Securities Australia Ltd Buy 1.253.880 1.20 000 UBS Securities Austr	-kun-14	LIBS Securities Australia Ltd	8.5	220	200	Ordinary
UBS Securities Australia Ltd Buy 1,165 120 UBS Securities Australia Ltd Buy 1,165 120 UBS Securities Australia Ltd Buy 1,2,605 2,017 UBS Securities Australia Ltd Buy 1,196 1,247 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 4,180 1,23 UBS Securities Australia Ltd Buy 4,180 1,24 UBS Securities Australia Ltd Buy 4,180 1,24 UBS Securities Australia Ltd Buy 1,176 1,2 UBS Securities Australia Ltd Buy 1,176 1,2 UBS Securities Australia Ltd Buy 1,26 1,2 UBS Securities Australia Ltd Buy 1,26 1,2 UBS Securities Australia Ltd Buy 1,26 1,2 UBS Securities Australia Ltd Buy 1,253,880 1,2,5 UBS Securities Australia Ltd Buy 1,253,880 1,2,5 UBS Securities Australia Ltd Sell <td>-kin-14</td> <td>LiBS Securities Australia Ltd</td> <td>Berry</td> <td>377</td> <td>2,</td> <td>ording)</td>	-kin-14	LiBS Securities Australia Ltd	Berry	377	2,	ording)
UBS Securities Australia Ltd Buy 3,820 3,93 UBS Securities Australia Ltd Buy 19,605 2017 UBS Securities Australia Ltd Buy 6,075 6,25 UBS Securities Australia Ltd Buy 6,075 6,27 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,205 1,24 UBS Securities Australia Ltd Buy 1,205 1,24 UBS Securities Australia Ltd Buy 1,205 1,23 UBS Securities Australia Ltd Buy 1,255 840 1,705 UBS Securities Australia Ltd Buy 1,255 840 1,706 1,70 UBS Securities Australia Ltd Sell Sell 35,605 3,7021 1,725 1,70 1,70 1,70 1,70 1,70	-lun-14	UBS Securities Australia Itd	R. V	1 186	02,	Ordinary
UBS Securities Australia Ltd Buy 19,050 2,017 UBS Securities Australia Ltd Buy 12,121 1,247 UBS Securities Australia Ltd Buy 6,075 6,075 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 4,180 4,30 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,206 1,23 UBS Securities Australia Ltd Buy 1,206 1,27 UBS Securities Australia Ltd Buy 1,255,880 1,29,000 UBS Securities Australia Ltd Buy 1,255,880 1,29,000 UBS Securities Australia Ltd Buy 1,255,880 1,29,000 UBS Securities Australia Ltd Sell 25,815 1,775 UBS Securities Australia Ltd Sell 35,805 1,725 UBS Securities Australia Ltd Buy 1,255,805 1,700 UBS Securities Aus	-Jun-14	UBS Securities Australia Ltd	Brix	2820	303	Ordinary
UBS Securities Australia Ltd Buy 12,721 1,247 UBS Securities Australia Ltd Buy 6,075 6,275 6,275 UBS Securities Australia Ltd Buy 2,4,096 2,479 1,23 UBS Securities Australia Ltd Buy 2,4,096 1,23 UBS Securities Australia Ltd Buy 4,180 1,23 UBS Securities Australia Ltd Buy 1,176 1,23 UBS Securities Australia Ltd Buy 1,176 1,24 UBS Securities Australia Ltd Buy 1,260 1,21 UBS Securities Australia Ltd Buy 1,260 1,21 UBS Securities Australia Ltd Buy 1,250 1,24 UBS Securities Australia Ltd Buy 1,253,880 1,25,000 UBS Securities Australia Ltd Buy 1,253,880 1,25,000 UBS Securities Australia Ltd Buy 1,253,880 1,25,000 UBS Securities Australia Ltd Sell 25,147 7,825 UBS Securities Australia Ltd Sell 25,147 7,825 </td <td>-lim-14</td> <td>LIRS Securities Australia Ltd</td> <td>Bry</td> <td>10 605</td> <td>2012</td> <td>Grana</td>	-lim-14	LIRS Securities Australia Ltd	Bry	10 605	2012	Grana
UBS Securities Australia Ltd Buy 6,075 6,275 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 2,4796 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,176 1,13 UBS Securities Australia Ltd Buy 1,205 1,24 UBS Securities Australia Ltd Buy 1,205 1,24 UBS Securities Australia Ltd Buy 1,255 840 UBS Securities Australia Ltd Buy 1,255 840 UBS Securities Australia Ltd Buy 1,255 840 UBS Securities Australia Ltd Sell 2,59,477 (5,5770) UBS Securities Australia Ltd Sell 35,806 (3,725) UBS Securities Australia Ltd Buy 75,741 7,825 UBS Securities Australia Ltd Buy 1,375 (3,625) UBS Securities Australia Ltd		1 IDC Convision Australia 14d	600	500,50	710,7	Crainery
UBS Securities Australia Ltd Buy 5,075 1,39 1,33 1,39 1,33 1,39 1,33 1,39 1,33 <	1	I IDC Commission Australia tad	and a	12,121	1,247	Ordinary
UBS Securities Australia Ltd Buy 1,196 1,29 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,23 UBS Securities Australia Ltd Buy 1,196 1,13 UBS Securities Australia Ltd Buy 1,176 1,17 UBS Securities Australia Ltd Buy 1,176 1,17 UBS Securities Australia Ltd Buy 1,176 1,17 UBS Securities Australia Ltd Buy 1,253,880 1,20 UBS Securities Australia Ltd Buy 2,58,839 1,20 UBS Securities Australia Ltd Buy 1,253,880 1,20 UBS Securities Australia Ltd Buy 2,58,339 2,6,70 UBS Securities Australia Ltd Sell 3,5,06 (3,722) UBS Securities Australia Ltd Buy 75,41 7,825 UBS Securities Australia Ltd Sell 25,30 (3,699) UBS Securities Australia Ltd Sell 25,314 7,70 UBS Securities Australia Ltd </td <td>1</td> <td>LIDE Commission A contract</td> <td>na c</td> <td>6/0,0</td> <td>579</td> <td>Ordinary</td>	1	LIDE Commission A contract	na c	6/0,0	579	Ordinary
UBS Securities Australia Ltd Buy 4,186 4,79 UBS Securities Australia Ltd Buy 1,196 123 UBS Securities Australia Ltd Buy 1,196 123 UBS Securities Australia Ltd Buy 1,176 121 UBS Securities Australia Ltd Buy 1,205 1,24 UBS Securities Australia Ltd Buy 1,205 1,21 UBS Securities Australia Ltd Buy 1,253,880 1,20 UBS Securities Australia Ltd Buy 1,253,880 1,20 UBS Securities Australia Ltd Buy 1,253,880 1,29,000 UBS Securities Australia Ltd Sell 2,59,437 (5,770) UBS Securities Australia Ltd Sell 35,80 (7,725) UBS Securities Australia Ltd Sell 37,021 (3,825) UBS Securities Australia Ltd Sell 37,021 (3,834) UBS Securities Australia Ltd Sell 37,021 (3,659) UBS Securities Australia Ltd Sell 35,1021 (3,659) UBS Securitie	-Jul - 14	Upo Securicies Australia Ltd	Buy	961,1	123	Ordinary
UBS Securities Australia Ltd Buy 1,196 103 UBS Securities Australia Ltd Buy 4,180 103 UBS Securities Australia Ltd Buy 1,196 123 UBS Securities Australia Ltd Buy 1,040 107 UBS Securities Australia Ltd Buy 1,040 107 UBS Securities Australia Ltd Buy 8,266 840 UBS Securities Australia Ltd Buy 8,266 840 UBS Securities Australia Ltd Buy 8,266 840 UBS Securities Australia Ltd Buy 1,253,880 129,000 UBS Securities Australia Ltd Buy 1,253,880 129,000 UBS Securities Australia Ltd Sell 256,314 7,874 1,775 UBS Securities Australia Ltd Sell 35,40 7,574 7,825 UBS Securities Australia Ltd Buy 75,741 7,825 UBS Securities Australia Ltd Sell 259,147 7,659 UBS Securities Australia Ltd Sell 25,312 7,701	MD-14	OBS Securities Australia Ltd	Buy	24,096	2,479	Ordinary
UBS Securities Australia Ltd Bby 4,180 430 UBS Securities Australia Ltd Bby 1,176 121 UBS Securities Australia Ltd Buy 1,76 121 UBS Securities Australia Ltd Buy 1,205 124 UBS Securities Australia Ltd Buy 1,265 840 UBS Securities Australia Ltd Buy 1,253,880 120 UBS Securities Australia Ltd Buy 1,253,880 1,200 UBS AG London Branch Buy 1,253,880 1,270 UBS AG London Branch Sell 258,339 1,727 UBS AG London Branch Sell 3,702 1,723 UBS Securities Australia Ltd Sell 35,806 (1,384) UBS Securities Australia Ltd Sell 35,702 1,725 UBS Securities Australia Ltd Sell 259,147 (1,725 UBS Securities Australia Ltd Sell 37,021 (3,635) UBS Securities Australia Ltd Sell 25,741 (2,6770) UBS Securities Australia Ltd <th< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1,196</td><td>123</td><td>Ordinary</td></th<>	-Jun-14	UBS Securities Australia Ltd	Buy	1,196	123	Ordinary
UBS Securities Australia Ltd Buy 1,196 123 UBS Securities Australia Ltd Buy 1,706 124 UBS Securities Australia Ltd Buy 1,040 107 UBS Securities Australia Ltd Buy 1,040 107 UBS Securities Australia Ltd Buy 1,25,880 120 UBS Securities Australia Ltd Buy 1,25,880 129,000 UBS Securities Australia Ltd Sell 256,539 26,770 UBS Securities Australia Ltd Sell 35,800 (1,384) UBS Securities Australia Ltd Sell 35,7021 (3,722) UBS Securities Australia Ltd Sell 35,7021 (38,992) UBS Securities Australia Ltd Sell 37,021 (38,992) UBS Securities Australia Ltd Sell 37,021 (38,992) UBS Securities Australia Ltd Sell 37,01 (26,700) UBS Securities Australia Ltd Sell 37,01 (26,700) UBS Securities Australia Ltd Buy 16,744 1,725 UBS Sec	-Jun-14	UBS Securities Australia Ltd	Buy	4,180	430	Ordinary
UBS Securities Australia Ltd Bby 1,176 121 UBS Securities Australia Ltd Bby 1,209 1,24 UBS Securities Australia Ltd Bby 1,206 121 UBS Securities Australia Ltd Bby 6,266 840 UBS Securities Australia Ltd Bby 1,253,880 129,000 UBS Securities Australia Ltd Bby 2,58,339 26,770 UBS Securities Australia Ltd Sell 259,147 7,825 UBS Securities Australia Ltd Sell 35,806 (1,324) UBS Securities Australia Ltd Sell 35,806 (1,324) UBS Securities Australia Ltd Sell 35,906 (1,324) UBS Securities Australia Ltd Sell 259,147 7,574 7,574 UBS Securities Australia Ltd Sell 25,314 7,574 7,575 UBS Securities Australia Ltd Sell 25,314 7,574 7,585 UBS Securities Australia Ltd Sell 25,317 2,539 2,530 UBS Securities Australia Ltd Sell	-Jun-14	UBS Securities Australia Ltd	Buy	1,196	123	Ordinary
UBS Securities Australia Ltd BM 1,205 124 UBS Securities Australia Ltd Buy 1,205 107 UBS Securities Australia Ltd Buy 1,265 840 UBS Securities Australia Ltd Buy 1,253.880 129,000 UBS AG London Branch Buy 1,253.880 129,000 UBS AG London Branch Buy 258,339 26,770 UBS AG London Branch Buy 258,339 27,700 UBS AG London Branch Sell 258,339 27,700 UBS Securities Australia Ltd Sell 35,806 37,701 UBS Securities Australia Ltd Sell 37,021 78,22 UBS Securities Australia Ltd Sell 37,021 78,22 UBS Securities Australia Ltd Sell 25,741 78,25 UBS Securities Australia Ltd Sell 26,755 1,725 UBS Securities Australia Ltd Sell 26,755 1,725 UBS Securities Australia Ltd Sell 26,756 1,725 UBS Securities Australia Ltd <td< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1,176</td><td>121</td><td>Ordinary</td></td<>	-Jun-14	UBS Securities Australia Ltd	Buy	1,176	121	Ordinary
UBS Securities Australia Ltd Bby 1,040 107 UBS Securities Australia Ltd Buy 8,266 840 UBS Securities Australia Ltd Buy 1,253,880 129,000 UBS Securities Australia Ltd Buy 1,253,880 129,000 UBS Securities Australia Ltd Sell 258,835 26,770 UBS Securities Australia Ltd Sell 13,398 (1,384) UBS Securities Australia Ltd Sell 35,800 (3,722) UBS Securities Australia Ltd Sell 35,800 (3,722) UBS Securities Australia Ltd Sell 35,800 (3,722) UBS Securities Australia Ltd Sell 35,900 (3,722) UBS Securities Australia Ltd Sell 35,900 (3,722) UBS Securities Australia Ltd Sell 35,900 (3,722) UBS Securities Australia Ltd Sell 16,754 (7,723) UBS Securities Australia Ltd Sell 22,312 (2,639) UBS Securities Australia Ltd Sell 22,312 (2,305)	-Jun-14	UBS Securities Australia Ltd	Buy	1,205	124	Ordinary
UBS Securities Australia Ltd Buy 6,256 1,776 121 UBS Securities Australia Ltd Buy 6,266 120000 UBS Securities Australia Ltd Buy 256,539 76,770 UBS AG London Branch Buy 256,539 26,770 UBS AG London Branch Seal 256,539 26,770 UBS AG London Branch Seal 258,539 26,770 UBS Securities Australia Ltd Seal 35,805 (1,384) UBS Securities Australia Ltd Seal 35,805 (2,698) UBS Securities Australia Ltd Seal 26,255 (2,698) UBS Securities Australia Ltd Seal 26,255 (2,698) UBS Securities Australia Ltd Buy 87,701 (5,698) UBS Securities Australia Ltd Enter Into Derivative Q N/A 3,835	Jun-14	UBS Securities Australia Ltd	Buy	1,040	107	Ordinary
UBS Securities Australia Ltd Buy 1,253,880 1,253,880 1,250,000 UBS AG London Barch Buy 1,253,880 1,29,000 UBS AG London Barch Sell 258,439 26,770 UBS AG London Barch Sell 258,439 26,770 UBS AG London Barch Sell 259,437 26,770 UBS Securities Australia Ltd Sell 37,721 27,723 UBS Securities Australia Ltd Sell 37,021 (1,824) UBS Securities Australia Ltd Sell 37,021 (8,922) UBS Securities Australia Ltd Sell 37,021 (8,922) UBS Securities Australia Ltd Sell 37,021 (8,922) UBS Securities Australia Ltd Sell 25,341 (7,524 UBS Securities Australia Ltd Sell 26,255 (2,599) UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Enter Into Derivative C N/A 3,835	Jun-14	UBS Securities Australia Ltd	Buy	1,176	121	Ordinary
UBS Securities Australia Ltd Buy 1,253,880 129,000 UBS AG London Branch BM 258,539 26,770 UBS AG London Branch Sell 259,447 257,700 UBS AG London Branch Sell 259,447 237 UBS Securities Australia Ltd Sell 13,398 (1,384) UBS Securities Australia Ltd Sell 35,000 (1,324) UBS Securities Australia Ltd Sell 37,021 (3,722) UBS Securities Australia Ltd Sell 37,021 (3,892) UBS Securities Australia Ltd Sell 37,021 (3,892) UBS Securities Australia Ltd Sell 35,447 (26,770) UBS Securities Australia Ltd Sell 36,447 (26,770) UBS Securities Australia Ltd Sell 22,312 (2,598) UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Enter Into Derivative NA 3,835 UBS Securities Australia Ltd Enter Into Derivative NA 3,835 <t< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>8.266</td><td>840</td><td>Ordinary</td></t<>	-Jun-14	UBS Securities Australia Ltd	Buy	8.266	840	Ordinary
UBS AG London Branch Buy 258,539 26,770 UBS AG London Branch Sell 259,147 (26,770) UBS Securities Australia Ltd Stock borrowed 13,398 (1,384) UBS Securities Australia Ltd Sell 13,398 (1,384) UBS Securities Australia Ltd Sell 37,7021 7825 UBS Securities Australia Ltd Sell 37,7021 (36,292) UBS Securities Australia Ltd Sell 259,147 (26,770) UBS Securities Australia Ltd Sell 259,147 (26,770) UBS Securities Australia Ltd Buy 16,754 1,725 UBS Securities Australia Ltd Buy 16,754 1,725 UBS Securities Australia Ltd Buy 16,754 1,725 UBS Securities Australia Ltd Buy 841,220 87,045 UBS Securities Australia Ltd Enter Into Derivative Quarterial Ltd 87,001 (3,305) UBS Securities Australia Ltd Enter Into Derivative Quarterial Ltd 87,701 (596)	-Jun-14	UBS Securities Australia Ltd	Buy	1,253,880	129.000	Ordinary
UBS AG London Branch Sell 259,147 (26,770) UBS Securities Australia Ltd Stock borrowed NA 232 UBS Securities Australia Ltd Sell 13,398 (1,384) UBS Securities Australia Ltd Sell 35,806 (3,722) UBS Securities Australia Ltd Sell 37,021 78,741 7,825 UBS Securities Australia Ltd Sell 259,17 (88,592) UBS Securities Australia Ltd Sell 25,701 (88,592) UBS Securities Australia Ltd Sell 26,255 (7.70) UBS Securities Australia Ltd Sell 26,255 (7.70) UBS Securities Australia Ltd Sell 26,255 (7.54) (7.25) UBS Securities Australia Ltd Sell 22,312 (2,305) (7.305) UBS Securities Australia Ltd Enter Into Derivative C NA 3,835 UBS Securities Australia Ltd Enter Into Derivative C NA 3,835 UBS Securities Australia Ltd Enter Into Derivative C 7,701 (596)	-Jun-14	UBS AG London Branch	Buy	258,539	26.770	Ordinary
UBS Securities Australia Ltd Stock borrowed NA 232 UBS Securities Australia Ltd Sell 13,398 (1,384) UBS Securities Australia Ltd Sell 35,600 (3,722) UBS Securities Australia Ltd Sell 75,741 7,825 UBS Securities Australia Ltd Sell 16,754 1,725 UBS Securities Australia Ltd Sell 22,312 (2,699) UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Enter Into Derivative NA 3,835 UBS Securities Australia Ltd Enter Into Derivative NA 3,835 UBS Securities Australia Ltd Enter Into Derivative NA 2,770 UBS Securities Australia Ltd Sell 5,770 (5,56	-Jun-14	UBS AG London Branch	Sell	259,147	(26.770)	Ordinary
Sell 13,398 (1,384) Sell 35,806 (3,722) Buy 75,741 (38,922) Sell 377,021 (38,922) Buy 16,754 (1,25) Sell 259,147 (26,770) Sell 26,255 (2,599) Enter Into Derivative 22,312 (2,305) Enter Into Derivative N/A 3,835 Enter Into Derivative N/A 3,835 Sell 25,701 (596) Sell 25,701	4-Jun-14	UBS Securities Australia Ltd	Stock borrowed	¥2	232	Ordinary
UBS Securities Australia Ltd Sell 35,806 (3,722) UBS Securities Australia Ltd Buy 75,41 7,825 UBS Securities Australia Ltd Sell 37,021 (88,992) UBS Securities Australia Ltd Sell 259,147 (26,770) UBS Securities Australia Ltd Buy 16,754 1,725 UBS Securities Australia Ltd Buy 841,220 87,045 UBS Securities Australia Ltd Sell 22,312 (2,899) UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Enter Into Derivative Granties Australia Ltd 3,835 UBS Securities Australia Ltd Enter Into Derivative Granties Australia Ltd 200 UBS Securities Australia Ltd Enter Into Derivative Granties Australia Ltd 27,312 (2,305)	-Jun-14	UBS Securities Australia Ltd	Sell	13 398	(1 384)	Ordinary
UBS Securities Australia Ltd Buy 75,741 7,825 UBS Securities Australia Ltd Sell 75,741 7,825 UBS Securities Australia Ltd Sell 259,147 (26,770) UBS Securities Australia Ltd Bull 16,754 1,725 UBS Securities Australia Ltd Sell 26,255 (2,698) UBS Securities Australia Ltd Buy 841,220 (2,698) UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Finter Into Derivative N/A 3,835 UBS Securities Australia Ltd Enter Into Derivative N/A 3,835 UBS Neathly Management Australia Ltd Sell 5770 (596)	Jun-14	UBS Securities Australia Ltd	Iss	35,806	(3 722)	Ordinary
UBS Securities Australia Ltd Sell 377/021 (88,922) UBS Securities Australia Ltd Sell 259,147 (26,770) UBS Securities Australia Ltd Buy 1,725 (26,898) UBS Securities Australia Ltd Buy 841,220 87,045 UBS Securities Australia Ltd Sell 26,235 (2,698) UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Enter Into Derivative G NA 3,835 UBS Securities Australia Ltd Enter Into Derivative G NA 3,835 UBS Securities Australia Ltd Enter Into Derivative G NA 3,835 UBS Securities Australia Ltd Enter Into Derivative G NA 2,300 UBS Securities Australia Ltd Enter Into Derivative G 5,770 (5,565)	-Jun-14	UBS Securities Australia Ltd	Buv	75.741	7 875	Ordinary
UBS Securities Australia Ltd Sell 259,147 (6,770) UBS Securities Australia Ltd Buy 16,734 1,725 UBS Securities Australia Ltd Sell 26,245 1,725 UBS Securities Australia Ltd Buy 841,220 87,048 UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Enter Into Derivative C NA 3,835 UBS Securities Australia Ltd Enter Into Derivative C NA 3,835 UBS Securities Australia Ltd Enter Into Derivative C NA 3,835 UBS Nealth Management Australia Ltd Sen 27,710 (596)	-Jun-14	UBS Securities Australia Ltd	la _S	120 225	(38 992)	Ordinan
UBS Securities Australia Ltd Buy 16,754 1,725 UBS Securities Australia Ltd Sell 26,255 (2,689) UBS Securities Australia Ltd Buy 841,220 (2,689) UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Enter Into Derivative C N/A 3,835 UBS Securities Australia Ltd Enter Into Derivative C N/A 3,835 UBS Neathly Management Australia Ltd Sell 57701 (596)	kn-14	UBS Securities Australia Ltd	3	259 147	(06,25)	Ording
UBS Securities Australia Ltd Sell 26,355 2,698 UBS Securities Australia Ltd Buy 841,220 87,045 UBS Securities Australia Ltd Sell 22,312 (2,305) UBS Securities Australia Ltd Finter Into Derivative C NA 3,835 UBS Securities Australia Ltd Enter Into Derivative C NA 2,00 UBS Securities Australia Ltd Enter Into Derivative C NA 200 UBS Securities Australia Ltd S770 (596)	-lun-14	UBS Securities Australia Ltd	a a	15.75	1 725	ording)
UBS Securities Australia Ltd Sell Co.253 14,0369 UBS Securities Australia Ltd Buy 841,220 87,0489 UBS Securities Australia Ltd Sell (2,305) 87,056 UBS Securities Australia Ltd Enter into Derivative Q N/A 3,835 UBS Securities Australia Ltd Enter into Derivative Q N/A 3,835 UBS Securities Australia Ltd Enter into Derivative Q N/A 3,835 UBS Neath Management Australia Ltd Sell 5,770 (596)	11	I DC Committee Australia 144	c all	10,701	(002.0)	Organiary
UBS Securities Australia Ltd Self Enter Into Derivative USS Securities Australia Ltd Self Enter Into Derivative USS Securities Australia Ltd Enter Into Derivative USS Securities Australia Ltd Enter Into Derivative USS Securities Australia Ltd Enter Into Derivative USS Securities Australia Ltd Self Enter Into Derivative USS Securities Australia Ltd Self Enter Into Derivative USS STATO	July 14	OBS Securities Australia Lid	Jac.	cc7'q7	(2,698)	Ordinary
UBS Securities Australia Ltd Enter into Derivative Q NA 3,835 UBS Veralities Australia Ltd Enter into Derivative Q NA 3,835 UBS Veralith Management Australia Ltd Sell 2770 (596)	1	UBS Securities Australia Ltd	any 	077',188	87,045	Ordinary
UBS Securities Australia Lid Enter into Derivative d N/A 3,835 UBS Securities Australia Enter into Derivative d N/A 200 UBS Neath Management Australia Ltd Sell 5,770 (596)	Jun-14	UBS Securities Australia Ltd	Sel	22,312	(2,305)	Ordinary
UBS Securities Australia Ltd Enter into Derivative (NVA 200 UBS Wealth Management Australia Ltd Sell 5,770 (596)	Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	3,835	Options
UBS Wealth Management Australia Ltd Sell 5-770	Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (VΑ	200	Ontions

23-Jun-14	1 DC Commission A metallic Lead	, 0	10,		
3-140-14	I IRS Securities Australia Ltd	Buy	13/	151	Ordinary
110-14	I IRS Securities Australia I to	And a	2/4/1	131	Ordinary
23 Jun 14	I IDO Constitute Australia Lad	900	402',	740	Ordinary
# :	Obs Securities Australia Ltd	gnà	3,695	3/9	Ordinary
53-JUN-14	UBS Securities Australia Ltd	Buy	7,118	730	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	975	9	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	410	42	Ordinary
lun-14	UBS Securities Australia Ltd	Buy	5,801	265	Ordinary
23-Jun-14	UBS Securities Australia Ltd	Buy	156	16	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	168	95	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	7 032	777	Ordinary
3-kin-14	I JBS Securities Australia Ltd	BIN	1 032	30,5	Ording)
1,	1 IDC Countries Australia 1+d	900	200,1	3 [O Colling
	The state of the s	ána s	27475	/20	Ordinary
1	Obs Securities Australia Ltd	Buy	190,82	7,881	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	11,974	1,230	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	21,057	2,163	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	21,436	2,202	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	1,110	114	Ordinary
un-14	UBS Securities Australia Ltd	Buy	1.996	205	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	2 473	25.4	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	9170	042	Ordinan
3-hin-14	I IBS Securities Australia (14	Bire	1 733	170	Ording)
1, 1, 1	LIDE Committee Australia Lea	śna	25/1	9/2	Ordinary
1	OBS Seconnes Australia Liu	and .	150	34	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	69/	6/	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	69/	79	Ordinary
un-14	UBS Securities Australia Ltd	Buy	769	79	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	692	79	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	692	6/	Ordinary
3-Jun-14	UBS Securities Australia ind	Stiv	769	70	Ordinary
3-1 in-14	I IBC Coourities Australia I to	2.6	2007	100	O'C'III
	THE STATE OF THE PARTY OF THE P	60	60/	õ	Cidinary
3-Jun-14	UBS Securities Australia Ltd	Buy	69/	6	Ordinary
UD-14	UBS Securities Australia Ltd	Buy	769	79	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	17,941	1,842	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	2,240	230	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	16.753	1.720	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	526	27	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	765	61	Ordinan
3-1-in-14	I IBC Concrition Australia I to	(-) d	1 1		0.00
	DI SI TI VI TI LI LI LI LI LI LI LI LI LI LI LI LI LI	, na	6/1	0 1	Ordinary
5-Jun-14	Obs securities Australia Ltd	Buy	49	Λ.	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	89	7	Ordinary
un-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
2 lin 14	I IBS Sociutios Australia (14)	Braz	000	1 0	Ordinary Continue
	I DC Counting Australia Ltd	Ana	000	2	Cidiriary
23-JUIT-14	Upp Securities Australia Ltd	Buy	66/	79	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	692	79	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	88	o	Ordinary
14	URS Securities Australia I td	Bry	101	,	Ordinan
23-hin-14	URC Countition Australia 14d	D. C.	800	7 00	Ordinaly
	COD Securities Australia Lid	ouy.	900	8	Cluriary
3-Jun-14	UBS Securities Australia Ltd	gany	88	5	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	101	-	Ordinary
Jn-14	UBS Securities Australia Ltd	Buy	61	2	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Birv	818	28	Ordinary
2 to 14	Del citation original Co. DOI -	6	0 0	5	
1 1 1	Upo Complete Australia Liu	Buy	000	7 (Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	789	81	Ordinary
un-14	UBS Securities Australia Ltd	Buy	88	6	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	085	30	Ordinary
	LAI Contribut Australia Lat	, , ,	200	3	, in the
-1nin-c	ODD DECURIES AUSTRAINA LIG	had	64	c	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buy	799	82	Ordinary
3-jun-14	UBS Securities Australia Ltd	Buy	88	6	Ordinary
3-Jun-14	UBS Securities Australia Ltd	Buv	10	-	Ordinary
1-Jun-14	UBS Securities Australia Ltd	Buv	67	u	Ordinary
	ממים מים מים מים מים	(na	C+		
7,77		!			
41-10C-C7	UBS Securities Australia Ltd	Buy	682	18	Ordinary

24-Jun-14 24-Jun-14 24-Jun-14	Opp wegin individual en Mustidia Liu				
Jun-14	I IRS Wealth Management Australia (14	Je	17,433	(1,103)	Ordinary
-Jun-14	tiBS Wealth Management Australia 11d	والم	11647	(1,703)	Ordinary
	I IRC Mosth Management American 14d	100	11,047	(102,1)	Guarary
	ODD WESTER INSTITUTION AUSTRALIA	lac .	7/7//	(1,/84)	Ordinary
24-JUN-14	UBS Wealth Management Australia Ltd	Sell	17,456	(1,803)	Ordinary
24-Jun-14	UBS wealth Management Australia Ltd	Sell	11,763	(1,215)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,231	(1,160)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	16,962	(1,752)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	1,162	(120)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	5,083	(222)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	17.340	(1.791)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	11 589	(1 197)	Ordina
4-lim-14	I IRS Wealth Management Australia Ltd	Coll	204.51	(1 200)	i i i
24-11-14	I ISC Month Management Australia Ltd	loc d	104,77	(1,790)	Counsily
	ODS Wealth Mariagerherit Australia Ltd	ilac	0///5	(969)	Ordinary
-1	U85 Wealth Management Australia Ltd	Self	1,462	(151)	Ordinary
- [UBS Wealth Management Australia Ltd	Seli	4,685	(484)	Ordinary
- 1	UBS Wealth Management Australia Ltd	Buy	1,094	113	Ordinary
	UBS Wealth Management Australia Ltd	Sell	20,967	(2,166)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	1.133	(117)	Ordinary
	UBS Wealth Management Australia Ltd	Sell	1.142	(118)	Ordinan
ı	UBS Wealth Management Australia Ltd	las	1133	(711)	Ordinary.
24-1 m-14	I IBC Wealth Management Australia 1th	100	1 4 4 5	(011)	or in a
1 1 1 1 1	The state of the s	ner.	741,1	(0 1)	Ordinary
1	Obs wealth Management Australia Ltd	Seil	2,790	(864)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Seil	5,770	(296)	Ordinary
-Jun-14	UBS Wealth Management Australia Ltd	Sell	16,720	(1,727)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	6,061	(929)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	16,855	(1,741)	Ordinary
4-lun-14	I IRS Wealth Management Australia 14d	Coll	11 570	(1 106)	Ordings
4-1 10-14	I IRC Moath Management Australia 14d	100	003 0	1001,11	Cidinary
1 1 1 1 1 1 1	Libra Wearth Wallage Hellt Australia Ltd	śna "	860'7	/07	Ordinary
+1	ODS Wealth Management Australia Ltd	Sell	5,790	(268)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	1,714	(177)	Ordinary
-Jun-14	UBS Wealth Management Australia Ltd	Sell	5,790	(208)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	2,790	(865)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Seli	17 398	(1 797)	Ordinary
4-lin-14	ILIBS Wealth Management Australia 11d	190	062.3	(609)	Ordinary
24-lun-14	I IBC Woolth Management Australia Ltd	100	11,000	(1 100)	or or or or
14	I IDO Committee Australia (+a)		600,11	(00)	Oldinary -
24-Jul-14	Obs securities Australia Ltd	Buy	4,070	420	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	9,225	952	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	5,601	578	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	775	08	Ordinary
24-lun-14	LIBS Securities Australia Ltd	Blay	109 575	11 308	Ording
1	1 DC Committee Australia 14d	3.00	000.07	006,1	Cidinal
1	Obs securities Australia Liu	Buy	nno'ns	2,160	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	295	28	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	629	89	Ordinary
4-Jun-14	UBS Securities Australia Ltd	Buy	12.645	1,305	Ordinary
24-Jun-14	UBS Securities Australia Ltd	NI.S	11 134	1 140	Ordinary
24-hm-14	I IBS Societies Australia Ltd	0.00	1000	21.7	O di iai
1 to 1 to 1 to 1 to 1 to 1 to 1 to 1 to	1 10 C C	(60	1000		O Gilliary
1 .	Day Securities Australia Ltu	, and	13,023	1,344	Ordinary
24-JUD-14	UBS Securities Australia Ltd	Buy	514	23	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	1,076	111	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	920	64	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	10,766	1,111	Ordinary
4-Jun-14	UBS Securities Australia Ltd	Sec.	14.535	1.500	Ordinary
24-lim-14	I IRS Securities Australia Ltd	Bin	0000	100	Ordingo
	I DC Commission Australia Ltd	(ha	2,020	167	Cignary
+1-1mr-+2	Obs securities Australia Ltd	pon	30,524	3,150	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	1,938	200	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	28,101	2,900	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	1,134	117	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buv	707	73	Ordinary
Bin-14	I JBS Securities Australia 14d	× 18	1 071	513) region
	I IDC Commission of the second commission of t	, and	1/6'#	0 1	Colorary
4-JUI-14	Upo securities Australia Ltd	Buy	6,539	155	Ordinary
4-Jun-14	UBS Securities Australia Ltd	Buy	6,744	969	Ordinary
- 1	UBS Securities Australia Ltd	Buy	106,590	11,000	Ordinary
4-Jun-14	UBS Securities Australia Ltd	Buy	696	9	Ordinary
ŀ	UBS Securities Australia Ltd	Buy	19 380	0000	١
Т	ODS Securities Australia בנכ	l l l	100000		/450

1 1 1 1 2	I DO 187 In the second second second			1000	
4-lun-14	UBS Wealth Management Australia Ltd	- E	272 71	(137)	Ordinary
1-14	UBS Wealth Management Australia Ltd	les	11,637	(1,202)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,707	(1,829)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	5,567	(575)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,378	(1,795)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Buy	1,703	175	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,598	(1,198)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	9 3	6,138	(634)	Ordinary
7-100-14	LIBC Worlds Management Australia Ltd	133	2,55	(610)	Ordinary
24-Jun-14	I IRS Wealth Management Australia Ltd	i i i	17 195	(2,249)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	1 de 2	780	(507)	Ordinary
n-14	UBS Wealth Management Australia Ltd	Sell	11.618	(1,200)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	5,760	(262)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,608	(1,199)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,272	(1,784)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,637	(1,202)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	5,790	(298)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Nei!	16,894	(1,745)	Ordinary
24-Jun-14	Libe Wealth Management Australia Ltd	Sell Sell	17,340	(16/1)	Ordinary
4-hin-14	LIBS Wealth Management Australia (14	Jeg Jeg	17.485	(202,1)	Ordinary
24-lun-14	UBS Wealth Management Australia Ltd	<u> </u>	5.741	(503)	Ordinary
14	UBS Wealth Management Australia Ltd	38	6.157	(636)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,436	(1,801)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Seli	6,061	(626)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,446	(1,802)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Seli	12,470	(1,288)	Ordinary
4-jun-14	UBS Wealth Management Australia Ltd	Sell	17,427	(1,800)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,763	(1,215)	Ordinary
n-14	UBS Wealth Management Australia Ltd	Sell	11,589	(1,197)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	24,117	(2,491)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,618	(1,200)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Self	11,589	(1,197)	Ordinary
24-Jun-14	UBS wealth Management Australia Ltd	Sell	5,741	(593)	Ordinary
24-Jun-14	USS Wearth Marragement Australia Ltd	Sell Sell	11,214	(8//1)	Ordinary
24-Jun-14	USS Wealth Management Australia Ltd	lles lles	11,802	(612,1)	Ordinary
1 1	LIDS Wealth Mastagement Australia Ind	liac liac	17 308	(200,1)	Ordinary
24-Jun-14	II IRS Wealth Management Australia Ltd	Sell Sell	10 01	(750.5)	Ordinary
12	IURS Wealth Management Australia Ltd	3	5,867	(606)	Ordinary
12	UBS Wealth Management Australia Ltd	is S	790,5	(598)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sei	17,436	(1.801)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sel	17.291	(1.786)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	5.819	(601)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	17.446	(1,802)	Ordinary
n-14	UBS Wealth Management Australia Ltd	les.	11 627	(1 201)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	1,317	(136)	Ordinary
24-jun-14	UBS Wealth Management Australia Ltd	Sell	17,349	(1,792)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Setl	17,165	(1,773)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,444	(1,182)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,214	(1,778)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,378	(1,795)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	17,127	(1,769)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	5,799	(665)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	11,627	(1,201)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sell	5,790	(865)	Ordinary
4-Jun-14	UBS Wealth Management Australia Ltd	Sett	11,782	(1,217)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	4,550	(470)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	5,896	(609)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	2,760	(262)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	2,760	(565)	Ordinary
24-Jun-14	UBS Wealth Management Australia Ltd	Sell	2,790	(298)	Ordinary
ŀ					
4	UBS Wealth Management Australia Ltd	Sell	17.640	(1.822)	Ordinary

UBS Securities Australia Ltd	Buy	349	36	Ordinary
JBS Securities Australia Ltd	Buy	552	57	Ordinary
JBS Securities Australia Ltd	Buv	573	54	Ordinary
JBS Securities Australia Ltd	A S	116	12	Ordinary
UBS Securities Australia I1d	Bir	271	30	ordinally ordinally
UBS Securities Australia Ltd	8.19	194	02	Ordinary
istralia i td	2.0	1066	110	Ordinary
UBS Securities Australia Itd	Si S	920,1	1 253	Ordinary
UBS Securities Australia Ltd	Buv	504	52	Ordinany
ustralia I td	Bury	212	32	Ordings
UBS Securities Australia I to	Biro	170 11	1 545	Ordinary
istralia I to	Biss	670	202	O'CHILLING.
introlia 14d	600	0000	200	Oldinary
TIDE Cochistics Assets 144	bay.	4,690	404	Ordinary
Ustralia LIO	ing	595	991	Ordinary
Obsisted Australia Ltd	Buy	3,8/6	400	Ordinary
Securities Australia Ltd	Buy	41,202	4,252	Ordinary
UBS Securities Australia Etd	Buy	5,339	551	Ordinary
UBS Securities Australia Ltd	Buy	63,741	6,578	Ordinary
UBS Securities Australia Ltd	Buy	3,876	400	Ordinary
ıstralia Ltd	Buy	5,775	296	Ordinary
stralia Ltd	Buv	7.219	745	Ordinary
UBS Securities Australia Ltd	Buy	15,620	1,612	Ordinary
UBS Securities Australia Ltd	Bliv	20 400	300 6	Ordinary
UBS Securities Australia I to	Bin	100,000	2000	Ordinary
of city	600	13,024	400	Ordinary
IBC Consistion Australia 14d	600	4,555	004	Cidilaly
Straile Ltd	ing	91-18	941	Crainary
obs securities Australia Lid	Buy	540	99	Ordinary
UBS Securities Australia Ltd	Buy	11,318	1,168	Ordinary
stralia Ltd	Buy	6,570	678	Ordinary
tralia Ltd	Buy	5,339	551	Ordinary
tralia Ltd	Buy	6,744	969	Ordinary
UBS Securities Australia Ltd	Buy	2,818	167	Ordinary
stralia Ltd	Buy	174	18	Ordinary
JBS Securities Australia Ltd	Buy	1,948	201	Ordinary
UBS Securities Australia Ltd	Buy	9,700	1,001	Ordinary
UBS Securities Australia Ltd	Buy	16,667	1,720	Ordinary
UBS Securities Australia Ltd	Buy	707	73	Ordinary
stralia Ltd	Buy	40,349	4,164	Ordinary
stralia Ltd	Buy	696	100	Ordinary
UBS Securities Australia Ltd	Buy	63.741	6.578	Ordinary
UBS Securities Australia I td	Si.v	3.876	VOV.	Ordinan
UBS Securities Australia Itd	Biv	5 330	12	Ordinary
IBS Securities Australia I to	A Contract of the Contract of	2000	090	Ordinary
Straila Lit	Suy.	005,5	308	Ordinary
Obs securities Australia Ltd	BUY	9,215	951	Ordinary
ustralia Ltd	Buy	281	29	Ordinary
JBS Securities Australia Ltd	Buy	10,165	1,049	Ordinary
JBS Securities Australia Ltd	Buy	591	19	Ordinary
UBS Securities Australia Ltd	Buy	13,469	1,390	Ordinary
JBS Securities Australia Ltd	Buy	12,219	1,261	Ordinary
stralia Ltd	Buy	21,347	2,219	Ordinary
UBS Securities Australia Ltd	Buy	11,640	1,210	Ordinary
UBS Securities Australia Ltd	Buy	42,713	4,440	Ordinary
stralia Ltd	Buy	13.688	1.414	Ordinary
UBS Securities Australia Ltd	Buv	17.763	1.835	Ordinary
UBS Securities Australia Ltd	Buy	610	63	Ordinary
stralia I td	Bliv	2 972	307	, de dipo
Securities Australia I to	Biry	1 701	i a	Ording
IBS Sequenties Australia 14d	100	7000	000	Crumialy
IRC Counties Australia 14d	600	10,030	4,300	Cruiriary
100 Counties Australia Ltd	l gang	002,11	1,104	Ordinary
Stiglia LtO	anà.	38,788	4,005	Ordinary
JBS Securities Australia Ltd	Buy	15,420	1,593	Ordinary
Securities Australia Ltd	Buy	2,449	253	Ordinary
UBS Securities Australia Ltd	Buy	13,223	1.366	Ordinary
Part of the last				5
ODS SECURIES AUSTRAIN LTD	Buy	4,830	499	Ordinary

28,644 2,956 Ordinary	693	1,600		4,725	382		1,358	1,236	200		4118		3	770	244	October 1934	/ 000	Socio 2339 Ordinary	3.766	121	400	1,977 204 Ordinary	1,000	204	325	63	400	3,651	261	16,667 1,720 Ordinary	2,920	2,180 225 Ordinary	55	13	2,424	204	223		17	69		2,5,5	1	27.006 2.787 Ordinally	3 473	137	L	285	13	620		4,525 467 Ordinary			35	136 the Ordinary	3.4		01.0		2.053	
Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buck	Buy	Buy	200	, and	, and	A COLO	a a	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy.	Buy	And a	Bug.	200	Buv	Buy	buy B.s.	Buy	B. P.	Buy	Riv	652	86	Buy								
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	10BS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IRS Securities Australia I I d	HIDS Committee Australia Ltd	LIBS Societies Australia 14d	TIRS Societies Australia 144	I IBS Societies Australia #td	UBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIRS Securities Australia Ltd	URS Securities Australia 14d	LIBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Ups Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	LIBS Securities Australia 14d	UBS Securities Australia Etd	UBS Securities Australia 11d		UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd
24-Jun-14	24-Jun-14	24-Jun-14	24-JUN-14	24-Jun-14	24-Jun-14	24-Jun-14	24-Jun-14	24-Jun-14			ı	24-Jun-14	ı		24 100-14	1	ı	24-lun-14		24-Jun-14				24-Jun-14		24-Jun-14	- 1	- 1	24-Jun-14	ı	ı	24-Jun-14	ı	24-Jun-14	П	ı	24-lun-14	24-lun-14	24-Jun-14	24-Jun-14	24-Jun-14	ı	- 1		- 1	24-Jun-14	24-Jun-14	24-JUN-14	24-Jun-14	24-Jun-14	24-lun-14	24-Jun-14	24-inn-14		24-Jun-14	24-Jun-14 24-Jun-14						

Septembre Andread Led Byay 13444 13456 1360 Ordensy School Led BOS Securities Andread Led Byay 4,555 5,00 Ordensy School Led BOS Securities Andread Led Byay 4,655 5,00 Ordensy School Led BOS Securities Andread Led Byay 4,655 5,00 Ordensy School Led BOS Securities Andread Led Byay 1,117 5,100 Ordensy School Led BOS Securities Andread Led Byay 1,62,17 5,00 Ordensy School Led BOS Securities Andread Led Byay 1,62,17 5,00 Ordensy School Led BOS Securities Andread Led Byay 1,62,17 1,73 Ordensy School Led BOS Securities Andread Led Byay 1,62 1,73 Ordensy School Led BOS Securities Andread Led Byay 1,62 0,73 1,13 0,74 School Led BOS Securities Andread Led Byay 1,62 0,74 1,13 0,74	5-Jun-14					
USS Securities Australia LID Buy 6,177 5,00 USS Securities Australia LID Buy 6,177 5,00 USS Securities Australia LID Buy 1,117 1,107 USS Securities Australia LID Buy 1,627 1,627 USS Securities Australia LID Buy 1,628 1,628 USS Securities Australia LID <td< td=""><td>110</td><td>UBS Securities Australia Ltd</td><td>Su.</td><td>13,944</td><td>1,436</td><td>Ordinary</td></td<>	110	UBS Securities Australia Ltd	Su.	13,944	1,436	Ordinary
USS Securities Australia LID BM 19.7 10.5 5.00 USS Securities Australia LID BM 19.4 19.420 5.00 USS Securities Australia LID BMy 19.420 1.00 USS Securities Australia LID BMy 1.655.5 1.00 USS Securities Australia LID BMy 1.655.0 1.70 USS Securities Australia LID BMy 1.655.0 1.70 USS Securities Australia LID BMy 1.655.0 1.70 USS Securities Australia LID BMy 4.07 1.655.0 USS Securities Australia LID BMy 1.05 1.00 USS Securities Australia LID BMy 1.05 1.00 USS Securities Australia LID BMy 2.03 1.05 USS Securities Australia LID BMy 2.03 1.05 USS	17	I IDS Counties Australia Ltd	eug eug	4,855	200	Ordinary
658 Securities Australia LID Buy 11,117 10,00 1058 Securities Australia LID Buy 11,117 12,00 1058 Securities Australia LID Buy 15,00 12,00 1058 Securities Australia LID Buy 16,00 17,11 105 Securities Australia LID Buy 16,00 17,21 105 Securities Australia LID Buy 16,00 17,22 105 Securities Australia LID Buy 16,00 17,22 105 Securities Australia LID Buy 16,00 17,23 105 Securities Australia LID Buy 16,00 17,23 105 Securities Australia LID Buy 16,00 17,23 105 Securities Australia LID Buy 10,00 17,23 105 Securities Australia LID Buy 10,00 17,00 105 Securities Australia LID Buy 20,00 17,00 105 Securities Australia LID Buy 20,00 17,00 105 Securities Australia LID Buy 20,00 17,00 105 Securities Australia LID	17	I IBC Sequirities Australia Ind	sna one	71 10	050	Ordinary
URS Securities Australia Ltd Buy 19,420 URS Securities Australia Ltd Buy 19,420 URS Securities Australia Ltd Buy 15,271 URS Securities Australia Ltd Buy 15,271 URS Securities Australia Ltd Buy 15,271 URS Securities Australia Ltd Buy 4,655 URS Securities Australia Ltd Buy 4,655 URS Securities Australia Ltd Buy 4,675 URS Securities Australia Ltd Buy 1,723 URS Securities Australia Ltd Buy 1,723 URS Securities Australia Ltd Buy 1,724 URS Securities Australia Ltd Buy 1,725 URS Securities Australia Ltd Buy	1 1 m	Libs Securities Australia Ltd	Ang Park	48,550	000,5	Ordinary
URS Securities Australia Ltd Buy 16,547 16,537 URS Securities Australia Ltd Buy 16,671 1,721 URS Securities Australia Ltd Buy 16,571 1,721 URS Securities Australia Ltd Buy 1,675 1,722 URS Securities Australia Ltd Buy 1,675 1,723 URS Securities Australia Ltd Buy 1,073 1,18 URS Securities Australia Ltd Buy 1,073 1,11 URS Securities Australia Ltd Buy 1,073 1,1 URS Securities Australia Ltd Buy 1,073 1,1 URS Securities Australia Ltd Buy 2,03 1,0 URS Securities Australia Ltd Buy 2,03 1,0 URS Securities Australia Ltd Buy 2,03 1,0 URS Securities Australia Ltd Buy </td <td>1</td> <td>Liber Countries Australia Liber</td> <td>iio d</td> <td>/11/1</td> <td>(5) 1)</td> <td>Ordinary</td>	1	Liber Countries Australia Liber	iio d	/11/1	(5) 1)	Ordinary
USS Securities Australia Ltd Buy 15,876 USS Securities Australia Ltd Buy 15,876 USS Securities Australia Ltd Buy 15,731 USS Securities Australia Ltd Buy 15,723 USS Securities Australia Ltd Buy 15,723 USS Securities Australia Ltd Buy 15,724 USS Securities Australia Ltd Buy 10,724 USS Securities Australia Ltd Buy 10,725 USS Securities Australia Ltd Buy 10,727 USS Securities Australia Ltd Buy 10,727 USS Securities Australia Ltd Buy 10,727 USS Securities Australia Ltd Buy 20,727 USS Securities Australia Ltd <	100	Obsider Australia Lid	euy	19,420	2,000	Ordinary
Construction Autorials Ltd Buy 15,816 1,521 UBS Securities Autorials Ltd Buy 15,817 1,121 UBS Securities Autorials Ltd Buy 6,600 (670) UBS Securities Autorials Ltd Buy 6,600 (670) UBS Securities Autorials Ltd Buy 1,523 1,296 UBS Securities Autorials Ltd Buy 1,600 1,600 UBS Securities Autorials Ltd Buy 2,23 1,600 UBS Securities Autorials Ltd Buy 2,23 1,600 UBS Securities Autorials Ltd <td< td=""><td>- Inc</td><td>Ubs securities Australia Ltd</td><td>Buy</td><td>160,477</td><td>16,527</td><td>Ordinary</td></td<>	- Inc	Ubs securities Australia Ltd	Buy	160,477	16,527	Ordinary
UBS Securities Australia Ltd Buy 6,506 (570) UBS Securities Australia Ltd Buy 6,506 (670) UBS Securities Australia Ltd Buy 6,506 (670) UBS Securities Australia Ltd Buy 1,523 1,766 UBS Securities Australia Ltd Buy 1,523 1,766 UBS Securities Australia Ltd Buy 1,038 1,156 UBS Securities Australia Ltd Buy 1,038 1,156 UBS Securities Australia Ltd Buy 1,038 1,156 UBS Securities Australia Ltd Buy 1,038 1,038 UBS Securities Australia Ltd Buy 1,038 1,038 UBS Securities Australia Ltd Buy 2,239 1,038 UBS Securities Australia Ltd Buy 2,239 1,038 UBS Securities Australia Ltd Buy 2,030 1,03 UBS Securities Australia Ltd Buy 2,03 1,03 UBS Securities Australia Ltd Buy 2,03 1,03 UBS Securities Australia Ltd Bu	100	USS SECURITIES AUSTRAINS LTG	Buy	15,876	1,635	Ordinary
UBS Securities Australia Ltd Bluy 6,851 5,134 UBS Securities Australia Ltd Bluy 6,2323 1,296 UBS Securities Australia Ltd Bluy 1,596 4,203 1,296 UBS Securities Australia Ltd Bluy 1,596 1,596 1,596 1,596 UBS Securities Australia Ltd Bluy 1,534 1,1,596 1,596	-Jun-14	UBS Securities Australia Ltd	Buy	16,711	1,721	Ordinary
URS Securities Australia Ltd Stell 6,500 (670) URS Securities Australia Ltd Buy 2,239 (670) URS Securities Australia Ltd Buy 4,678 17,060 URS Securities Australia Ltd Buy 4,678 17,060 URS Securities Australia Ltd Buy 10,344 11,063 URS Securities Australia Ltd Buy 10,368 10,368 URS Securities Australia Ltd Buy 2,37 10 URS Securities Australia Ltd Buy 2,39 1,37 URS Securities Australia Ltd Buy 2,39 1,37 URS Securities Australia Ltd Buy 2,30 1,48 URS Securities Australia Ltd Buy 2,30 1,50 URS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Buy	49,851	5,134	Ordinary
UBS Securities Australia Ltd Bly 12,333 2,296 UBS Securities Australia Ltd Bly 1,696 4,20 LBS Securities Australia Ltd Bly 1,696 4,20 LBS Securities Australia Ltd Bly 1,734 1,156 LBS Securities Australia Ltd Bly 30,247 4,20 LBS Securities Australia Ltd Bly 30,247 31,15 LBS Securities Australia Ltd Bly 30,247 31,15 LBS Securities Australia Ltd Bly 30,247 31,15 LBS Securities Australia Ltd Bly 30,247 30,38 LBS Securities Australia Ltd Bly 30,27 30,30 LBS Securities Australia Ltd Bly 2,33 30,30 LBS Securities Australia Ltd Bly 2,33 30,30 LBS Securities Australia Ltd Bly 2,34 30,30 LBS Securities Australia Ltd Bly 2,34 30,30 LBS Securities Australia Ltd Bly 2,32 30,30 LBS Securities Australia Ltd <th< td=""><td>Jun-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>905'9</td><td>(670)</td><td>Ordinary</td></th<>	Jun-14	UBS Securities Australia Ltd	Sell	905'9	(670)	Ordinary
UBS Securities Australia Ltd Buy 10,666 UBS Securities Australia Ltd Buy 4,078 17,667 UBS Securities Australia Ltd Buy 1,934 1,065 UBS Securities Australia Ltd Buy 1,033 1,035 UBS Securities Australia Ltd Buy 10,338 1,038 UBS Securities Australia Ltd Buy 10,388 1,038 UBS Securities Australia Ltd Buy 10,388 1,038 UBS Securities Australia Ltd Buy 10,388 1,037 1,1 UBS Securities Australia Ltd Buy 1,077 1,1 1,007 1,0 UBS Securities Australia Ltd Buy 2,237 2,0 2,0 UBS Securities Australia Ltd Buy 2,0 2,0 3,0 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Buy	22,323	2,299	Ordinary
UBS Securities Australia Ltd Buy 4,078 4	Jun-14	UBS Securities Australia Ltd	Buy	165,653	17,060	Ordinary
UBS Securities Australia Ltd Bluy 155 166 UBS Securities Australia Ltd Buy 19381 1,655 UBS Securities Australia Ltd Buy 10,7344 11,055 UBS Securities Australia Ltd Buy 10,734 11,055 UBS Securities Australia Ltd Buy 20,247 28 UBS Securities Australia Ltd Buy 49 20,247 28 UBS Securities Australia Ltd Buy 40 25 26 UBS Securities Australia Ltd Buy 26 27 26 UBS Securities Australia Ltd Buy 27,31 27 26 UBS Securities Australia Ltd Buy 20,32 27 27 26 UBS Securities Australia Ltd Buy 20,37 27,30 27 26 47 UBS Securities Australia Ltd Buy 20,37 27,37 27 28 UBS Securities Australia Ltd Buy 20,37 27,37 27 28 27 20 27 27 27 <td>-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>4,078</td> <td>420</td> <td>Ordinary</td>	-Jun-14	UBS Securities Australia Ltd	Buy	4,078	420	Ordinary
UBS Securities Australia Ltd Buy 107,344 1,1956 UBS Securities Australia Ltd Buy 107,344 1,1956 UBS Securities Australia Ltd Buy 30,247 3,115 UBS Securities Australia Ltd Buy 30,247 3,115 UBS Securities Australia Ltd Buy 10,283 10,285 UBS Securities Australia Ltd Buy 40 36 36 UBS Securities Australia Ltd Buy 40 37 10 UBS Securities Australia Ltd Buy 20 27 27 UBS Securities Australia Ltd Buy 20 20 30 UBS Securities Australia Ltd Buy 20 20 30 UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Buy	155	16	Ordinary
UBS Securities Australia Ltd Bluy 10.0534 11.055 UBS Securities Australia Ltd Buy 30.247 3.18 UBS Securities Australia Ltd Buy 30.247 3.18 UBS Securities Australia Ltd Buy 49 3.6 UBS Securities Australia Ltd Buy 49 5.2 UBS Securities Australia Ltd Buy 2.2 2.6 UBS Securities Australia Ltd Buy 2.2 2.6 UBS Securities Australia Ltd Buy 2.7 2.6 UBS Securities Australia Ltd Buy 2.0 2.0 UBS Securities Australia Ltd Buy 2.0 2	Jun-14	UBS Securities Australia Ltd	Buy	19,381	1,996	Ordinary
UBS Securities Australia Ltd Buy 30.247 3115 UBS Securities Australia Ltd Buy 20.247 3115 UBS Securities Australia Ltd Buy 7.00 3.23 1.03.28 UBS Securities Australia Ltd Buy 7.27 2.6 UBS Securities Australia Ltd Buy 2.7 2.6 UBS Securities Australia Ltd Buy 2.037 3.0 UBS Securities Australia Ltd Buy 2.037 3.0 UBS Securities Australia Ltd Buy 2.037 2.0 UBS Securities Australia Ltd Buy 2.037 2.0 UBS Securities Australia Ltd Buy 2.0 2.0 UBS Securities Australia Ltd Buy 2.0 2.0 UBS Securities Australia Ltd Buy 2.0 2.0 UBS Securities Australia Ltd Buy	-Jun-14	UBS Securities Australia Ltd	Buy	107,344	11,055	Ordinary
UBS Securities Australia Ltd Bby 222 228 UBS Securities Australia Ltd Bby 100,838 10,328 UBS Securities Australia Ltd Bby 49 11 UBS Securities Australia Ltd Bby 227 15 UBS Securities Australia Ltd Bby 243 13 UBS Securities Australia Ltd Bby 243 13 UBS Securities Australia Ltd Bby 243 13 UBS Securities Australia Ltd Bby 243 25 UBS Securities Australia Ltd Bby 39,06 403 UBS Securities Australia Ltd Bby 20,03 403 UBS Securities Australia Ltd Bby 20,03 400 403 UBS Securities Australia Ltd Bby 20,03 400 403 400 UBS Securities Australia Ltd Bby 20,03 400 400 100 UBS Securities Australia Ltd Bby 20,03 400 400 400 UBS Securities Australia Ltd Bby 20,00	-Jun-14	UBS Securities Australia Ltd	Buy	30,247	3,115	Ordinary
UBS Securities Australia Ltd Buy 100.838 10.385 UBS Securities Australia Ltd Buy 10.83 15.23 UBS Securities Australia Ltd Buy 12.833 1.523 UBS Securities Australia Ltd Buy 12.833 1.523 UBS Securities Australia Ltd Buy 272 26 UBS Securities Australia Ltd Buy 272 26 UBS Securities Australia Ltd Buy 2.03 38 UBS Securities Australia Ltd Buy 2.03 4.03 UBS Securities Australia Ltd Buy 2.03 4.03 UBS Securities Australia Ltd Buy 2.03 4.03 UBS Securities Australia Ltd Buy 2.03 2.03 UBS Securities Australia Ltd Buy 2	-Jun-14	UBS Securities Australia Ltd	Buy	272	28	Ordinary
UBS Securities Australia Ltd Buy 349 36 UBS Securities Australia Ltd Buy 405 349 36 UBS Securities Australia Ltd Buy 107 11 UBS Securities Australia Ltd Buy 255 10 UBS Securities Australia Ltd Buy 253 10 UBS Securities Australia Ltd Buy 253 10 UBS Securities Australia Ltd Buy 253 10 UBS Securities Australia Ltd Buy 25,314 25 UBS Securities Australia Ltd Buy 20,314 20 UBS Securities Australia Ltd Buy 20,314 20 UBS Securities Australia Ltd Buy 20,314 20 UBS Securities Australia Ltd Buy 20,314 403 UBS Securities Australia Ltd Buy 20,32 31 UBS Securities Australia Ltd Buy 20,27 40 UBS Securities Australia Ltd Buy 20,27 40 UBS Securities Australia Ltd Buy 20,27 <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>100,838</td> <td>10,385</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buy	100,838	10,385	Ordinary
UBS Securities Australia Ltd BMy 17.833 1.532 UBS Securities Australia Ltd BMy 12.833 1.532 UBS Securities Australia Ltd BMy 25.2 2.6 UBS Securities Australia Ltd BMy 97 1.0 UBS Securities Australia Ltd BMy 97 1.0 UBS Securities Australia Ltd BMy 22.310 2.300 UBS Securities Australia Ltd BMy 2.2310 2.300 UBS Securities Australia Ltd BMy 2.237 2.9 3 UBS Securities Australia Ltd BMy 2.2037 2.037 2.037 2.037 UBS Securities Australia Ltd BMy 2.2037 2.037 2.037 2.037 2.037 2.037 2.037 2.037 2.037 2.037 2.037 2.037 2.037 2.037	-Jun-14	UBS Securities Australia Ltd	Buy	349	36	Ordinary
UBS Securities Australia Ltd Buy 12.833 1,323 UBS Securities Australia Ltd Buy 252 26 UBS Securities Australia Ltd Buy 283 26 UBS Securities Australia Ltd Buy 283 36 UBS Securities Australia Ltd Buy 22.310 2.300 UBS Securities Australia Ltd Buy 2.037 1.5 UBS Securities Australia Ltd Buy 2.037 1.0 UBS Securities Australia Ltd Buy 2.037 2.037 UBS Securities Australia Ltd Buy 2.037 2.03 UBS Securities Australia Ltd Buy 2.037 2.037 UBS Securities Australia Ltd Buy <th< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>49</td><td>5</td><td>Ordinary</td></th<>	-Jun-14	UBS Securities Australia Ltd	Buy	49	5	Ordinary
UBS Securities Australia Ltd Buy 252 UBS Securities Australia Ltd Buy 287 1.0 UBS Securities Australia Ltd Buy 383 1.0 UBS Securities Australia Ltd Buy 383 9.1 UBS Securities Australia Ltd Buy 383 9.1 UBS Securities Australia Ltd Buy 2.23.10 2.300 UBS Securities Australia Ltd Buy 2.23.10 2.300 UBS Securities Australia Ltd Buy 2.23.10 2.300 UBS Securities Australia Ltd Buy 2.037 2.07 3.0 UBS Securities Australia Ltd Buy 2.037 2.0 3.0 UBS Securities Australia Ltd Buy 2.0 2.0	-Jun-14	UBS Securities Australia Ltd	Buy	12,833	1,323	Ordinary
UBS Securities Australia Ltd Buy 95 26 UBS Securities Australia Ltd Buy 243 26 UBS Securities Australia Ltd Buy 243 26 UBS Securities Australia Ltd Buy 264 15 UBS Securities Australia Ltd Buy 23,314 5,479 UBS Securities Australia Ltd Buy 2,314 5,479 UBS Securities Australia Ltd Buy 2,314 5,479 UBS Securities Australia Ltd Buy 2,314 5,479 UBS Securities Australia Ltd Buy 2,93 3 UBS Securities Australia Ltd Buy 2,93 3 UBS Securities Australia Ltd Buy 2,59 3 UBS Securities Australia Ltd Buy 2,59 3 UBS Securities Australia Ltd Buy 2,56 4,75 UBS Securities Australia Ltd Buy 2,56 4,25 4,25 UBS Securities Australia Ltd Buy 2,56 3,7 1,1 UBS Securities Australia Ltd B	Jun-14	UBS Securities Australia Ltd	Buy	107	11	Ordinary
UBS Securities Australia Ltd Bluy 249 10 UBS Securities Australia Ltd Buy 249 10 UBS Securities Australia Ltd Buy 369 38 UBS Securities Australia Ltd Buy 22,310 2,300 UBS Securities Australia Ltd Buy 2,2310 2,300 UBS Securities Australia Ltd Buy 2,037 2,03 UBS Securities Australia Ltd Buy 2,02 3 UBS Securities Australia Ltd Buy 2,03 3 UBS Securities Australia Ltd Buy 2,03 </td <td>-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>252</td> <td>26</td> <td>Ordinary</td>	-Jun-14	UBS Securities Australia Ltd	Buy	252	26	Ordinary
UBS Securities Australia Ltd BBy 369 35 UBS Securities Australia Ltd Bby 369 36 UBS Securities Australia Ltd Bby 369 37 UBS Securities Australia Ltd Bby 22,310 2,300 UBS Securities Australia Ltd Bby 2,037 2,10 UBS Securities Australia Ltd Bby 2,037 2,10 UBS Securities Australia Ltd Bby 2,037 2,03 UBS Securities Australia Ltd Bby 2,027 2,03 UBS Securities Australia Ltd Bby 2	4 I	UBS Securities Australia Ltd	Buy	97	10	Ordinary
UBS Securities Australia Ltd Bby 369 38 UBS Securities Australia Ltd Buy 176 13 UBS Securities Australia Ltd Buy 22,310 2,300 UBS Securities Australia Ltd Buy 22,310 2,479 UBS Securities Australia Ltd Buy 2,037 2,10 UBS Securities Australia Ltd Buy 2,037 20 UBS Securities Australia Ltd Buy 2,027 20 3 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	243	25	Ordinary
UBS Securities Australia Ltd BMy 126 13 UBS Securities Australia Ltd Buy 126 13 UBS Securities Australia Ltd Buy 22,310 2,300 UBS Securities Australia Ltd Buy 2,037 2,03 UBS Securities Australia Ltd Buy 2,037 2,07 UBS Securities Australia Ltd Buy 2,037 2,07 UBS Securities Australia Ltd Buy 2,037 2,07 UBS Securities Australia Ltd Buy 2,037 2,03 UBS Securities Australia Ltd Buy 2,037 2,03 UBS Securities Australia Ltd Buy 2,02 3 UBS Securities Australia Ltd Buy 2,02 <th< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>369</td><td>38</td><td>Ordinary</td></th<>	-Jun-14	UBS Securities Australia Ltd	Buy	369	38	Ordinary
UBS Securities Australia Ltd Bluy 148 91 UBS Securities Australia Ltd Buy 22,310 2,300 UBS Securities Australia Ltd Buy 22,314 2,300 UBS Securities Australia Ltd Buy 2,037 2,10 UBS Securities Australia Ltd Buy 2,037 2,10 UBS Securities Australia Ltd Buy 2,037 2,00 UBS Securities Australia Ltd Buy 2,037 2,00 UBS Securities Australia Ltd Buy 2,027 2,03 UBS Securities Australia Ltd Buy	-Jun-14	UBS Securities Australia Ltd	Buy	126	13	Ordinary
UBS Securities Australia Ltd Buy 22,310 15 UBS Securities Australia Ltd Buy 22,310 2,300 UBS Securities Australia Ltd Buy 53,146 5,479 UBS Securities Australia Ltd Buy 2037 210 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 20 3 UBS Securities Australia Ltd Buy 20 3 UBS Securities Australia Ltd Buy 4,26 47 UBS Securities Australia Ltd Buy 2,23 3 UBS Securities Australia Ltd Buy 2,26 3 UBS Securities Australia Ltd Buy 2,28 47 UBS Securities Australia Ltd Buy 2,28 47 UBS Securities Australia Ltd Buy 2,28 47 UBS Securities Australia Ltd Buy 2,34 2,39 UBS Securities Australia Ltd Buy 2,44 2,28	-Jun-14	UBS Securities Australia Ltd	Buy	883	50	Ordinary
UBS Securities Australia Ltd Buy 22,310 2,300 UBS Securities Australia Ltd Buy 2,231 2,479 UBS Securities Australia Ltd Buy 2,037 2,07 UBS Securities Australia Ltd Buy 2,037 2,00 UBS Securities Australia Ltd Buy 2,03 3 UBS Securities Australia Ltd Buy 2,0 3 UBS Securities Australia Ltd Buy 2,02 3 UBS Securities Australia Ltd Buy 2,75 3 UBS Securities Australia Ltd Buy 2,75 3 UBS Securities Australia Ltd Buy 2,75 3	Jun-14	UBS Securities Australia Ltd	Buy	146	15	Ordinary
UBS Securities Australia Ltd Buy 59.09 403 UBS Securities Australia Ltd Buy 5.0146 5.419 UBS Securities Australia Ltd Buy 2.037 2.10 UBS Securities Australia Ltd Buy 2.9 3 UBS Securities Australia Ltd Buy 2.9 3 UBS Securities Australia Ltd Buy 2.027 2.03 UBS Securities Australia Ltd Buy 2.04 2.05 UBS Securities Australia Ltd Buy 2.04 2.05 UBS Securities Australia Ltd Buy 2.04 2.05 UBS Securities Australia Ltd Buy 2.0	Jun-14	UBS Securities Australia Ltd	Buy	22,310	2,300	Ordinary
UBS Securities Australia Ltd Buy 59,146 5,499 UBS Securities Australia Ltd Buy 2,037 3 UBS Securities Australia Ltd Buy 2,99 3 UBS Securities Australia Ltd Buy 4,66 4,8 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 4,56 4,7 UBS Securities Australia Ltd Buy 2,027 7,8 UBS Securities Australia Ltd Buy 4,26 4,7 UBS Securities Australia Ltd Buy 2,027 7,8 UBS Securities Australia Ltd Buy 2,027 7,8 UBS Securities Australia Ltd Buy 2,027 2,9 3 UBS Securities Australia Ltd Buy 2,027 2,9 3 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 2,73 1,9 UBS Securities Australia Ltd Buy 2,44 2,5 UBS Securities Australia Ltd Buy<	-Jun-14	UBS Securities Australia Ltd	Buy	3,909	403	Ordinary
UBS Securities Australia Ltd Buy 2,037 210 UBS Securities Australia Ltd Buy 2,9 20 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 2,0 3 UBS Securities Australia Ltd Buy 4,5 4,9 UBS Securities Australia Ltd Buy 2,02 3 UBS Securities Australia Ltd Buy 2,02 3 UBS Securities Australia Ltd Buy 3,36 412 UBS Securities Australia Ltd Buy 3,36 412 UBS Securities Australia Ltd Buy 2,75 59 UBS Securities Australia Ltd Buy 5,75 59	Jun-14	UBS Securities Australia Ltd	Buy	53,146	5,479	Ordinary
UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 202 3 UBS Securities Australia Ltd Buy 2,027 709 UBS Securities Australia Ltd Buy 2,027 709 UBS Securities Australia Ltd Buy 2,027 209 UBS Securities Australia Ltd Buy 2,027 209 UBS Securities Australia Ltd Buy 2,027 20 UBS Securities Australia Ltd Buy 2,22 3 UBS Securities Australia Ltd Buy 2,24 2,27 UBS Securities Australia Ltd Buy 2,24 2,197 UBS Securities Australia Ltd Buy 2,24 2,197 UBS Securities Australia Ltd Buy 2,24 2,197	Jun-14	UBS Securities Australia Ltd	Buy	2,037	210	Ordinary
UBS Securities Australia Ltd Buy 1,940 200 UBS Securities Australia Ltd Buy 466 48 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 456 47 UBS Securities Australia Ltd Buy 2027 78 UBS Securities Australia Ltd Buy 4,264 47 UBS Securities Australia Ltd Buy 2,027 29 3 UBS Securities Australia Ltd Buy 3,266 412 UBS Securities Australia Ltd Buy 2,39 3 UBS Securities Australia Ltd Buy 2,99 3 UBS Securities Australia Ltd Buy 2,99 3 UBS Securities Australia Ltd Buy 2,752 59 UBS Securities Australia Ltd Buy 2,44 256 UBS Securities Australia Ltd Buy 2,44 256 UBS Securities Australia Ltd Buy 2,44 <	Jun-14	UBS Securities Australia Ltd	Buy	53	e	Ordinary
UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 757 78 UBS Securities Australia Ltd Buy 2,027 709 UBS Securities Australia Ltd Buy 4,258 439 UBS Securities Australia Ltd Buy 3,366 412 UBS Securities Australia Ltd Buy 3,366 412 UBS Securities Australia Ltd Buy 3,396 412 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 5,752 593 UBS Securities Australia Ltd Buy 5,752 5	Jun-14	UBS Securities Australia Ltd	Buy	1,940	200	Ordinary
UBS Securities Australia Ltd Buy 466 48 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 757 78 UBS Securities Australia Ltd Buy 757 78 UBS Securities Australia Ltd Buy 2,027 209 UBS Securities Australia Ltd Buy 4,258 439 UBS Securities Australia Ltd Buy 4,258 439 UBS Securities Australia Ltd Buy 3,966 412 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 2,94 3 UBS Securities Australia Ltd Buy 2,752 59 3 UBS Securities Australia Ltd Buy 2,744 222 9 3 UBS Securities Australia Ltd Buy 2,444 2,52 100 1,566 UBS Securities Australia Ltd Buy 2,444 2,52 100 1,566 UBS Securities Australia Ltd Buy 2,444 </td <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>58</td> <td>6</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buv	58	6	Ordinary
UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 757 78 UBS Securities Australia Ltd Buy 4,256 47 UBS Securities Australia Ltd Buy 2,027 209 UBS Securities Australia Ltd Buy 3,366 347 UBS Securities Australia Ltd Buy 3,996 412 UBS Securities Australia Ltd Buy 3,996 412 UBS Securities Australia Ltd Buy 2,99 3 UBS Securities Australia Ltd Buy 2,99 3 UBS Securities Australia Ltd Buy 2,752 59 UBS Securities Australia Ltd Buy 2,752 59 UBS Securities Australia Ltd Buy 2,731 1,19 UBS Securities Australia Ltd Buy 2,731 1,010 UBS Securities Australia Ltd Buy 2,731 1,010 UBS Securities Australia Ltd Buy 2,731	Jun-14	UBS Securities Australia Ltd	Buy	466	48	Ordinary
UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 456 47 UBS Securities Australia Ltd Buy 2,027 709 UBS Securities Australia Ltd Buy 2,027 39 UBS Securities Australia Ltd Buy 2,027 347 UBS Securities Australia Ltd Buy 3,366 347 UBS Securities Australia Ltd Buy 3,396 412 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 2,5 53 UBS Securities Australia Ltd Buy 2,5 53 UBS Securities Australia Ltd Buy 2,7 3 UBS Securities Australia Ltd Buy 5,752 59 UBS Securities Australia Ltd Buy 5,752 59 UBS Securities Australia Ltd Buy 5,311 2,197 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 5,325 549 <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Bity</td> <td>20</td> <td>i c</td> <td>Ordinap</td>	Jun-14	UBS Securities Australia Ltd	Bity	20	i c	Ordinap
UBS Securities Australia Ltd Buy 757 78 UBS Securities Australia Ltd Buy 456 47 UBS Securities Australia Ltd Buy 2,027 209 UBS Securities Australia Ltd Buy 4,258 439 UBS Securities Australia Ltd Buy 3,966 347 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2,444 252 UBS Securities Australia Ltd Buy 2,441 252 UBS Securities Australia Ltd Buy 2,331 1,566 UBS Securities Australia Ltd Buy 5,752 549 UBS Securities Australia Ltd Buy 5,725 <	Jun-14	UBS Securities Australia I td	Bity	220	7	Ordinary
UBS Securities Australia Ltd Buy 455 47 UBS Securities Australia Ltd Buy 4,256 47 UBS Securities Australia Ltd Buy 2,027 2.09 UBS Securities Australia Ltd Buy 3,366 347 UBS Securities Australia Ltd Buy 3,366 412 UBS Securities Australia Ltd Buy 2,996 412 UBS Securities Australia Ltd Buy 2,99 3 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 5,752 593 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 2,731 1,910 UBS Securities Australia Ltd Buy 2,1311 2,197 UBS Securities Australia Ltd Buy 2,325 549 UBS Securities Australia Ltd Buy 2,325 540 UBS Securities Australia Ltd Buy 2,325 </td <td>hin-14</td> <td>I IRS Securities Australia I trd</td> <td>8.5</td> <td>22</td> <td>202</td> <td>Ordinary</td>	hin-14	I IRS Securities Australia I trd	8.5	22	202	Ordinary
UBS Securities Australia Ltd Buy 2,027 249 UBS Securities Australia Ltd Buy 2,027 209 UBS Securities Australia Ltd Buy 3,966 347 UBS Securities Australia Ltd Buy 3,996 412 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 2,23 3 UBS Securities Australia Ltd Buy 2,744 2,52 UBS Securities Australia Ltd Buy 2,131 2,197 UBS Securities Australia Ltd Buy 5,722 59 UBS Securities Australia Ltd Buy 5,325	17	I IBC Countries Australia 14d	boy bear	15/	0 [Ordinary
UBS Securities Australia Ltd Buy 4,227 209 UBS Securities Australia Ltd Buy 4,256 33 UBS Securities Australia Ltd Buy 3,966 347 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2,444 25 UBS Securities Australia Ltd Buy 2,441 25 UBS Securities Australia Ltd Buy 5,752 549 UBS Securities Australia Ltd Buy 5,725 549 UBS Securities Australia Ltd Buy 5,725 549 UBS Securities Australia Ltd Buy 5,725	1	I IBC Contribute Authorities Library	, and	450	74/	Orginary
UBS Securities Australia Ltd Buy 4.258 4.39 UBS Securities Australia Ltd Buy 3,366 347 UBS Securities Australia Ltd Buy 3,396 412 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 5,752 593 UBS Securities Australia Ltd Buy 5,731 1,010 UBS Securities Australia Ltd Buy 5,735 549 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 2,335	- Inc	UBS Securities Australia Ltd	Buy	2,027	209	Ordinary
UBS Securities Australia Ltd Buy 3.366 3.7 UBS Securities Australia Ltd Buy 3.366 41.2 UBS Securities Australia Ltd Buy 1.319 136 UBS Securities Australia Ltd Buy 2.9 3 UBS Securities Australia Ltd Buy 2.9 3 UBS Securities Australia Ltd Buy 2.9 3 UBS Securities Australia Ltd Buy 2.444 2.52 UBS Securities Australia Ltd Buy 2.444 2.52 UBS Securities Australia Ltd Buy 2.731 1.010 UBS Securities Australia Ltd Buy 2.731 1.010 UBS Securities Australia Ltd Buy 5.725 5.49 UBS Securities Australia Ltd Buy 5.725 5.49 UBS Securities Australia Ltd Buy 5.225 5.49 UBS Securities Australia Ltd Buy 5.225 5.49 UBS Securities Australia Ltd Buy 5.225 5.49 UBS Securities Australia Ltd Buy 5.225<	Jun-14	UBS Securities Australia Ltd	Buy	4,258	439	Ordinary
UBS Securities Australia Ltd Buy 3,366 347 UBS Securities Australia Ltd Buy 3,996 412 UBS Securities Australia Ltd Buy 1,519 136 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 2,444 252 UBS Securities Australia Ltd Buy 2,1311 2,197 UBS Securities Australia Ltd Buy 9,13 1,00 UBS Securities Australia Ltd Buy 9,13 1,00 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 2,33	Jun-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
UBS Securities Australia Ltd Buy 3,996 412 UBS Securities Australia Ltd Buy 1,319 115 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 2,444 2,522 UBS Securities Australia Ltd Buy 2,131 2,197 UBS Securities Australia Ltd Buy 9,41 9,797 1,010 UBS Securities Australia Ltd Buy 9,41 1,586 UBS Securities Australia Ltd Buy 9,41 1,586 UBS Securities Australia Ltd Buy 3,235 549 UBS Securities Australia Ltd Buy 2,325 549 UBS Securities Australia Ltd Buy 2,325 549 UBS Securities Australia Ltd Buy 2,325 549 UBS Securities Australia Ltd Buy <td>Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>3,366</td> <td>347</td> <td>Ordinary</td>	Jun-14	UBS Securities Australia Ltd	Buy	3,366	347	Ordinary
UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 1,16 1,2 UBS Securities Australia Ltd Buy 2,9 3 UBS Securities Australia Ltd Buy 2,44 2,57 UBS Securities Australia Ltd Buy 2,444 2,52 UBS Securities Australia Ltd Buy 2,444 2,52 UBS Securities Australia Ltd Buy 2,731 1,010 UBS Securities Australia Ltd Buy 9,79 1,010 UBS Securities Australia Ltd Buy 5,225 5,49 UBS Securities Australia Ltd Buy 5,225 5,49 UBS Securities Australia Ltd Buy 5,225 5,49 UBS Securities Australia Ltd Buy 5,325 5,49 UBS Securities Australia Ltd Buy 1,319 1,36 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 1,319 1,319 UBS Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	3,996	412	Ordinary
UBS Securities Australia Ltd Buy 116 12 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2,752 593 UBS Securities Australia Ltd Buy 2,444 222 UBS Securities Australia Ltd Buy 2,441 2,56 UBS Securities Australia Ltd Buy 9,71,311 2,197 UBS Securities Australia Ltd Buy 9,71,311 2,197 UBS Securities Australia Ltd Buy 9,71,311 2,197 UBS Securities Australia Ltd Buy 9,71,311 1,56 UBS Securities Australia Ltd Buy 5,325 5,49 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 7,7 7,7 UBS Securities Australia Ltd Buy 7,9	-Jun-14	UBS Securities Australia Ltd	Buy	1,319	136	Ordinary
UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2752 593 UBS Securities Australia Ltd Buy 5,752 593 UBS Securities Australia Ltd Buy 2,444 252 UBS Securities Australia Ltd Buy 2,1,311 2,197 UBS Securities Australia Ltd Buy 9,797 1,010 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 2,335 549 UBS Securities Australia Ltd Buy 7,31 77 UBS Securities Australia Ltd Buy 7,32 7,01 UBS Securities Australia Ltd Buy 7,04	Jun-14	UBS Securities Australia Ltd	Buy	116	12	Ordinary
UBS. Securities Australia Ltd Buy 29 3 UBS. Securities Australia Ltd Buy 275 53 UBS. Securities Australia Ltd Buy 2,444 252 UBS. Securities Australia Ltd Buy 2,444 252 UBS. Securities Australia Ltd Buy 2,1311 2,1317 UBS. Securities Australia Ltd Buy 9,797 1,010 UBS. Securities Australia Ltd Buy 5,225 549 UBS. Securities Australia Ltd Buy 1,319 136 UBS. Securities Australia Ltd Buy 1,319 174 UBS. Securities Australia Ltd Buy 1,350 24 UBS. Securities Australia Ltd Buy 1,350 24 UBS. Securities Australia Ltd Buy	Jun-14	UBS Securities Australia Ltd	Buy	29	m	Ordinary
UBS Securities Australia Ltd Buy 5,752 93 UBS Securities Australia Ltd Buy 5,752 933 UBS Securities Australia Ltd Buy 2,444 258 UBS Securities Australia Ltd Buy 15,384 1,586 UBS Securities Australia Ltd Buy 2,1311 2,197 UBS Securities Australia Ltd Buy 941 97 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 2,335 549 UBS Securities Australia Ltd Buy 2,335 34 UBS Securities Australia Ltd Buy 2,335 34 UBS Securities Australia Ltd Buy 2,33 24 UBS Securities Australia Ltd Buy 7,7 77 UBS Securities Australia Ltd Buy 7,950 77 UBS Securities Australia Ltd Buy 7,042 7,761	Jun-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
UBS Securities Australia Ltd Buy 5,752 593 UBS Securities Australia Ltd Buy 2,444 222 UBS Securities Australia Ltd Buy 15,384 1,586 UBS Securities Australia Ltd Buy 21,311 2,197 UBS Securities Australia Ltd Buy 9,797 1,010 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 1,319 1,319 136 UBS Securities Australia Ltd Buy 2,335 549 UBS Securities Australia Ltd Buy 7,319 77 UBS Securities Australia Ltd Buy 7,31 7,74 UBS Securities Australia Ltd Buy 7,042 7,76	Jun-14	UBS Securities Australia Ltd	Buy	29	e	Ordinary
UBS Securities Australia Ltd Buy 2,444 252 UBS Securities Australia Ltd Buy 15,384 1,586 UBS Securities Australia Ltd Buy 2,1311 2,197 UBS Securities Australia Ltd Buy 9,797 1,010 UBS Securities Australia Ltd Buy 5,225 5,49 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 7,7 7,7 UBS Securities Australia Ltd Buy 7,950 201 UBS Securities Australia Ltd Buy 7,042 7,56	Jun-14	UBS Securities Australia Ltd	Buv	5 752	593	Ordinary
UBS Securities Australia Ltd Buy 15,884 1,586 UBS Securities Australia Ltd Buy 21,311 2,197 UBS Securities Australia Ltd Buy 9,737 1,010 UBS Securities Australia Ltd Buy 941 97 UBS Securities Australia Ltd Buy 5,325 5,49 UBS Securities Australia Ltd Buy 7,319 1,319 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 2,33 2,4 UBS Securities Australia Ltd Buy 7,7 77 UBS Securities Australia Ltd Buy 7,950 77 UBS Securities Australia Ltd Buy 7,042 7,56	Jun-14	UBS Securities Australia I td	N. S.	2 444	252	Ordinary.
UBS Securities Australia Ltd Buy 2 I,311 2,197 UBS Securities Australia Ltd Buy 9,797 1,010 UBS Securities Australia Ltd Buy 9,797 1,010 UBS Securities Australia Ltd Buy 5,325 5,49 UBS Securities Australia Ltd Buy 1,319 1,36 UBS Securities Australia Ltd Buy 747 77 UBS Securities Australia Ltd Buy 7,942 775 UBS Securities Australia Ltd Buy 7,042 726	Jun-14	IUBS Securities Australia Ltd	Biv	15 384	1 586	Ordinary
UBS Securities Australia Ltd Buy 9,797 1,010 UBS Securities Australia Ltd Buy 9,41 197 UBS Securities Australia Ltd Buy 5,225 5,49 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 233 24 UBS Securities Australia Ltd Buy 77 77 UBS Securities Australia Ltd Buy 1,950 77 UBS Securities Australia Ltd Buy 7,042 756 UBS Securities Australia Ltd Buy 7,042 756	Jun-14	IUBS Securities Australia Ltd	Buy	21 311	701.6	Ordinary
UBS Securities Australia Ltd Buy 941 97 UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 233 24 UBS Securities Australia Ltd Buy 77 77 UBS Securities Australia Ltd Buy 1,950 201 UBS Securities Australia Ltd Buy 7,042 756	Jun-14	UBS Securities Australia Ltd	A Section	797.6	1 010	Ordinary
UBS Securities Australia Ltd Buy 5,325 549 UBS Securities Australia Ltd Buy 1,319 136 UBS Securities Australia Ltd Buy 233 24 UBS Securities Australia Ltd Buy 747 77 UBS Securities Australia Ltd Buy 1,950 201 UBS Securities Australia Ltd Buy 7,042 726	Jun-14	UBS Securities Australia Itd	Bliv	941	26	Ordinan
UBS Securities Australia Ltd Buy 1,319 138 UBS Securities Australia Ltd Buy 233 24 UBS Securities Australia Ltd Buy 737 77 UBS Securities Australia Ltd Buy 747 77 UBS Securities Australia Ltd Buy 7,042 756	l.m-14	LIRS Securities Australia 14d	,	5225	200	Ordinary
1,219 1,21	11 m-14	LIBS Societies Australia 14d	S S S S S S S S S S S S S S S S S S S	1 210	136	Ordinary
UBS Securities Australia Ltd Buy 7.47 77 UBS Securities Australia Ltd Buy 1,950 201 UBS Securities Australia Ltd Buy 1,950 201 UBS Securities Australia Ltd Buy 7,042 726	1,01	1 BC Counties Australia 14d	(na)	565,1	130	Ordinally
VCO Securities Audatalia Ltd Boy 747 71 UBS Securities Australia Ltd Buy 1,950 201 UBS Securities Australia Ltd Buy 7,042 726	1 7	Libe counties Australia tea	, ind	233	77	Orginary
UBS Securities Australia Ltd Buy 7,950 201 1,950 201 202 726	1 1 1	Lings Securities Australia Ltd	Buy	/4/	//	Ordinary
UBS Securities Austraria Ltd Buy 726		Opp Securities Australia Ltd	, and	056,1	707	Croinary
	Jun-14	II IDC Congriting Auctralia 144	-			

T	UBS Securities Australia Ltd	Buy	086,61	7,054	Oldinary
24-Jun-14	UBS Securities Australia Ltd	Buy	7,909	817	Ordinary
T	UBS Securities Australia Ltd	Buy B	9,583	200	Ordinary
T	BS Securities Australia 11d	Rick	72071	1 759	Ordinary
24-lun-14	USS Securities Australia Itd	A Na	7 875	700	Ordinary
T	RS Securities Australia 14d	Brin	2,0,7	000	Ordinary
Ī	UBS Securities Australia Ltd	Bir	0299	089	Ordinary
	BS Securities Australia 11d	Riv	7,687	282	Ordinary
24-Jun-14	BS Securities Australia I td	Biv	9 583	Coo	Ordinaly
ı	UBS Securities Australia Ltd	Buv	136	14	Ordinary
	BS Securities Australia Ltd	Buv	25.758	7 661	Ordinary
Γ	UBS Securities Australia Ltd	Buv	8.247	852	Ordinary
ı	88 Securities Australia Ltd	Buy	30.474	3 143	Ordinary
24-Jun-14 U	BS Securities Australia Ltd	Buv	4.075	421	Ordinary
ł	UBS Securities Australia Ltd	Buv	14.520	1,500	Ordinary
ı	BS Securities Australia Ltd	Buv	9.496	186	Ordinary
Γ	UBS Securities Australia Ltd	Buy	8,247	852	Ordinary
24-Jun-14 UI	UBS Securities Australia Ltd	Buy	4,008	414	Ordinary
24-Jun-14 UI	UBS Securities Australia Ltd	Buy	3,098	320	Ordinary
	BS Securities Australia Ltd	Buy	964'6	186	Ordinary
Ī	UBS Securities Australia Ltd	Buy	2,255	233	Ordinary
Ī	UBS Securities Australia Ltd	Buy	3,446	356	Ordinary
24-Jun-14 UI	BS Securities Australia Ltd	Buy	14,723	1,521	Ordinary
- 1	UBS Securities Australia Ltd	Buy	3,098	320	Ordinary
- 1	BS Securities Australia Ltd	Buy	4,433	458	Ordinary
24-Jun-14 UI	UBS Securities Australia Ltd	Buy	2,652	274	Ordinary
- 1	8S Securities Australia Ltd	Buy	9,002	930	Ordinary
- 1	3S Securities Australia Ltd	Buy	7,599	785	Ordinary
╗	UBS Securities Australia Ltd	Buy	7,163	740	Ordinary
7	UBS Securities Australia Ltd	Buy	009	62	Ordinary
Ť	UBS Securities Australia Ltd	Buy	2,608	786	Ordinary
24-Jun-14 UE	JBS Securities Australia Ltd	Buy	1,113	115	Ordinary
┪	UBS Securities Australia Ltd	Buy	794	82	Ordinary
٦	UBS Securities Australia Ltd	Buy	862	88	Ordinary
	UBS Securities Australia Ltd	Buy	2,788	288	Ordinary
┪	UBS Securities Australia Ltd	Buy	3,320	343	Ordinary
24-Jun-14	UBS Securities Australia Ltd	Buy	3,678	380	Ordinary
	3S Securities Australia Ltd	Buy	862	68	Ordinary
٦	UBS Securities Australia Ltd	Buy	4,540	469	Ordinary
┪	UBS Securities Australia Ltd	Buy	39	4	Ordinary
	UBS Securities Australia Ltd	Buy	1,442	149	Ordinary
- 1	UBS Securities Australia Ltd	Buy	1,936	200	Ordinary
١	3S Securities Australia Ltd	Buy	5,992	619	Ordinary
24-Jun-14 UE	35 Securities Australia Ltd	Buy	11,219	1,159	Ordinary
- 1	UBS AG London Branch	Buy	273,768	28,247	Ordinary
	UBS AG London Branch	Sell	272,705	(28,247)	Ordinary
5-Jun-14 UE	UBS AG London Branch	Sell	409,273	(42,370)	Ordinary
Ĩ	JBS AG London Branch	Buy	410,868	42,370	Ordinary
Ī	JBS AG London Branch	Sell	680,682	(70,618)	Ordinary
Ē	35 AG London Branch	Buy	683,335	70.618	Ordinary
Ī	UBS Securities Australia Ltd	Stock returned	N/A	(232)	Ordinary
Ť	UBS Securities Australia Ltd	llaS	188 151	(19386)	Ordinary
Ť	UBS Securities Australia Ltd	Buv	201 404	20.817	Ordinary
Ť	UBS Securities Australia Ltd	Buy	44 044	4 549	Ordinary
25-Jun-14 UF	UBS Securities Australia Ltd	3	58.035	(5,995)	Ordinary
Τ	3S Securities Australia Ltd	Sell	11,633	(1,200)	Ordinary
Γ	UBS Securities Australia Ltd	Bliv	876 67Z	75.464	Ordinary
T	JBS Securities Australia I td	(S)	1 447 220	(149 181)	Ordinary
Ī	S Securities Australia I to	Birv	797.876	78.747	Ordinary
5-lin-14	IRS Sequenties Australia 14d	Buny	707,077 A10 866	42.247	Ordinary
Ť	20 Secondines Australia Lita	sna.	410,000	42,370	Ordinary
T	UBS Securities Australia Ltd	RUV	583,335	819,0/	Ordinary
Ť	UBS Securities Australia Ltd	cnter into Derivative C	NA	3,020	Options
Ť	JBS Global Asset Management	Sell	2,611	(269)	Ordinary
25-Jun-14 UB	UBS Securities Australia Ltd	Seli	27,858	(2,869)	Ordinary
	Securities Australia Ltd	Buv	1361	171	
				14	Ordinary

UBS Securities Australia Ltd Buy	25-Juli-14	I IDC Countition Australia (4)	, in	0.00	0.0	
1085 Securities Australia Ltd Buy Buy Buy Buy Bay Becurities Australia Ltd Buy Buy Buy Buy Bay Becurities Australia Ltd Buy Buy Buy Buy Bay Becurities Australia Ltd Buy Buy Buy Bay Becurities Australia Ltd Buy Buy Buy Buy Becurities Australia Ltd Buy Buy Buy Buy Becurities Australia Ltd Buy Buy Buy Buy Bay Becurities Australia Ltd Buy Buy Buy Buy Buy Becurities Australia Ltd Buy Buy Buy Buy Buy Becurities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	-lin-14	I IRS Securities Australia and	Buy	8,148	840	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy Buy Buy UBS Securities Australia Ltd Buy Buy Buy Buy <td>-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>18 537</td> <td>1,911</td> <td>Ordinary</td>	-Jun-14	UBS Securities Australia Ltd	Buy	18 537	1,911	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd <td< td=""><td>-Jun-14</td><td>UBS Securities Australia I td</td><td>Bliv</td><td>755'C1</td><td>5</td><td>Ordinary Crisis</td></td<>	-Jun-14	UBS Securities Australia I td	Bliv	755'C1	5	Ordinary Crisis
1.08.5 Securities Australia Ltd 1.08.5 Securities Australia Lt	-lun-14	1185 Securities Australia Ltd	60	OCO'T	200	Ordinally
1985 Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	-Jun-14	UBS Securities Australia Ltd	Birv	12,038	170,	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd <th< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>S S</td><td>89</td><td>7</td><td>Codinary</td></th<>	-Jun-14	UBS Securities Australia Ltd	S S	89	7	Codinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd <th< td=""><td>-lin-14</td><td>LIRS Securities Australia Ltd</td><td>2.00</td><td>999 /</td><td>, 0,</td><td>Oldinaly Oldinaly</td></th<>	-lin-14	LIRS Securities Australia Ltd	2.00	999 /	, 0,	Oldinaly Oldinaly
UBS Securities Australia Ltd UBS Securities A	-hin-14	ILIRS Securities Australia 11d	Buy	582.6	p c	or in a
UBS Securities Australia Ltd UBS Securities A	-Jun-14	UBS Securities Australia (11d	Birv	088 6	000	Ordinary
UBS Securities Australia Ltd UBS Securities A	-kin-14	LIBS Securities Australia I td	Bus	3 891	000	ordinal y
UBS Securities Australia Ltd UBS Securities A	-Jun-14	UBS Securities Australia 11d	Ba e	5,031	230 CE1	Cidinary
UBS Securities Australia Ltd UBS Securities A	-l.m-14	185 Securities Australia Ltd	Bray	2,040	500	Cidinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Australia Ltd Enter Into Devirative CUBS Securities Au	13-14	I IRS Countries Australia 14d	in d	00.00	023	Codiciany
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd <td< td=""><td>1</td><td>1 DC Coomition Australia Ltd</td><td>bod)</td><td>626,6</td><td>0/2</td><td>Orginary</td></td<>	1	1 DC Coomition Australia Ltd	bod)	626,6	0/2	Orginary
UBS Securities Australia Ltd Buy Buy UBS Securities Australia Ltd Buy Buy Buy UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy		Lab Commission Australia Liu	And a	97/	2	Ordinary
UBS Securities Australia Ltd Buy Buy Buy UBS Securities Australia Ltd Buy Buy Buy UBS Securities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into De	1	Library Australia Lid	Buy	34,047	3,510	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	1 1 1	Obs securities Australia Ltd	ans	0/8/89	7,100	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Into Derivative CUBS Securiti	1110	Obs securities Australia Ltd	Buy	/77'/	/45	Ordinary
UBS Securities Australia Ltd Buy Buy UBS Securities Australia Ltd Buy Buy Buy UBS Securities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Enter Into Devirative C UBS Securities Australia Ltd Enter I	11	Upo comition Australia LLO	had	0/6	201	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Buy Basecurities Australia Ltd Enter Into Derivative CUBS Securities Australia L	1 - 11	USS SECURITIES AUStrang LIG	Buy	29,149	3,005	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	-101-14	Obs securities Australia Ltd	Buy	5,345	551	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	-10t-14	Ubs securities Australia Ltd	Buy	1,125	116	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	100	Upo Securities Australia Ltd	huy	10,565	1,100	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Bus Securities Australia Ltd Buy Buy Bus Securities Australia Ltd Buy Buy Bus Securities Australia Ltd Buy Buy Bus Securities Australia Ltd Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Bus Securities Australia Ltd Buy Buy Bus Securities Australia Ltd Buy Buy Buy Bus Securities Australia Ltd Buy Buy Bus Buy Bus Bus Bus Bus Bus Bus Bus Bus Bus Bus	-Jun-14	UBS Securities Australia Ltd	Buy	6,858	707	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	Jun 14	UBS Securities Australia Ltd	Buy	7,223	745	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Basecurities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	Jun-14	UBS Securities Australia Ltd	Buy	87,901	9,062	Ordinary
UBS Securities Australia Ltd Buy Buy UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	-Jun-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	-Jun-14	UBS Securities Australia Ltd	Buy	14,579	1,503	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd <t< td=""><td>-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>13,890</td><td>1,432</td><td>Ordinary</td></t<>	-Jun-14	UBS Securities Australia Ltd	Buy	13,890	1,432	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	-Jun-14	UBS Securities Australia Ltd	Buy	20,894	2.154	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	-Jun-14	UBS Securities Australia Ltd	Buv	1.785	184	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Berivative C UBS Securities Australia Ltd Enter Into Berivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Berivative C UBS Securities Australia Ltd Enter Into Berivative C	-Jun-14	UBS Securities Australia ! td	Buy	7518	777	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	-Jun-14	UBS Securities Australia Ltd	Bis	32 135	2416	Crainary
USS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	-lin-14	I BS Securities Australia 14d	600	1 00.	0.4,7	O'CHINE O
USS Securities Australia Ltd Buy	1	1 Ibo Committee Australia Ltd	900	750,1 750,5	501	Ordinary
USS Securities Australia Ltd Buy	1	CDS Securities Australia Ltd	, and	3,036	313	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell 172 UBS Securities Australia Ltd Sell 18 UBS Securities Australia Ltd Buy 4 UBS Securities Australia Ltd Buy 4 UBS Securities Australia Ltd Buy 4 UBS Securities Australia Ltd Enter Into Derivative C Buy UBS Securities Australia Ltd Enter Into Derivative C C UBS Securities Australia Ltd Enter Into Derivative C C <t< td=""><td>1000</td><td>Obs securities Australia Ltd</td><td>gnà</td><td>4,850</td><td>200</td><td>Ordinary</td></t<>	1000	Obs securities Australia Ltd	gnà	4,850	200	Ordinary
UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Sell UBS Securities Australia 1td Sell UBS Securities Australia 1td Buy UBS Securities Australia 1td Buy UBS Securities Australia 1td Enter Into Derivative C	- Inn-14	UBS Securities Australia Ltd	Buy	4,850	200	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia L	-Jun-14	UBS Securities Australia Ltd	Buy	5,335	250	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia L	-Jun-14	UBS Securities Australia Ltd	Buy	1,086	112	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C </td <td>-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>7,227</td> <td>745</td> <td>Ordinary</td>	-Jun-14	UBS Securities Australia Ltd	Buy	7,227	745	Ordinary
UBS Securities Australia Ltd Buy Buy UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	-Jun-14	UBS Securities Australia Ltd	Buy	369	38	Ordinary
UBS Securities Australia Ltd Buy Buy UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	-Jun-14	UBS Securities Australia Ltd	Buv	11 999	1 237	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Ente	-Jun-14	UBS Securities Australia Ltd	Siv	417	EV	Company
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter I	-lin-14	LIBS Securities Australia Ltd	Bliv	06		diam's
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C	-Hin-14	LIBS Securities Australia Ltd	Biry	0,0		diam's
UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	1	I IDO Contribut A total and a set of	Śno d	D : .	7	Ordinary
UBS Securities Australia Ltd Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	101-14	Obs Securities Australia Ltd	Buy	4,355	449	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Buy UBS Securities Australia Ltd Sell Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative CUBS Securities Australia Ltd Enter Enter Into Derivative CUBS Securities Australia Ltd Enter Enter Into Derivative CUBS Securities Australia Ltd Enter En	101	Upo Securities Australia Ltd	Buy	4,355	449	Ordinary
UBS Securities Australia Ltd Sell Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell Buy UBS Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Buy UBS Securities Australia Ltd Sell Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	- Jun - 14	UBS Securities Australia Ltd	Buy	5,044	520	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sall UBS Securities Australia Ltd Sall UBS Securities Australia Ltd Sall UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Ernter Into Derivative C UBS Securities Australia Ltd Ernter Into Derivative C UBS Securities Australia Ltd Ernter Into Derivative C UBS Securities Australia Ltd Ernter Into Derivative C UBS Securities Australia Ltd Ernter Into Derivative C UBS Securities Australia Ltd Ernter Into Derivative C UBS Securities Australia Ltd Ernter Into Derivative C UBS Securities Australia Ltd Ernter Into Derivative C UBS Securities Australia Ltd Ernter Into Derivative C	Jun-14	UBS Securities Australia Ltd	Self	080'69	(7,084)	Ordinary
UBS Securities Australia Ltd Sell Sell UBS Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Buy UBS Securities Australia Ltd Buy Securities Australia Ltd Buy Securities Australia Ltd Sell Enter Into Derivative CUBS Securities Australia Ltd Enter Into Enter	-Jun-14	UBS Securities Australia Ltd	Buy	182,080	18,714	Ordinary
UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Bay UBS Securities Australia Ltd Sell UBS Securities Australia Ltd Enter Into Derivative C	-Jun-14	UBS Securities Australia Ltd	Buy	20,351	2,094	Ordinary
UBS Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	-Jun-14	UBS Securities Australia Ltd	Sell	129.385	(13,307)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	-Jun-14	UBS Securities Australia Ltd	Sell	291.819	(30,000)	Ordinary
UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Sell Securities Australia Ltd Buy UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	-lin-14	LIBS Securities Australia Ltd	9	12,610	(1 200)	
USS Securities Australia Ltd Sell Securities Australia Ltd Buy Securities Australia Ltd Buy Execurities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	1 1	LIDS Counties Australia Ltd	d	019,21	(000'1)	Orginary
UBS Securities Australia Ltd Bry Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	lin 1	I IBC Countries Australia Ltd	ing 3	44,0/3	4,023	Orginary
UBS Securities Australia Ltd Enter Into Derivative (1985 Securities Australia Ltd Ent	1 100	Obsideration Australia Ltd) Sell	89,541	(512/6)	Ordinary
UBS Securities Australia Ltd Enter Into Dernvative C UBS Securities Australia Ltd Enter Into Dernvative C UBS Securities Australia Ltd Enter Into Dernvative C UBS Securities Australia Ltd Enter Into Dernvative C UBS Securities Australia Ltd Enter Into Dernvative C UBS Securities Australia Ltd Enter Into Dernvative C UBS Securities Australia Ltd Enter Into Dernvative C UBS Securities Australia Ltd Enter Into Dernvative C	Jun-14	UBS Securities Australia Ltd		88,553	9,117	Ordinary
UBS Securities Australia Ltd Enter into Derivative C UBS Securities Australia Ltd Enter into Derivative C UBS Securities Australia Ltd Enter into Derivative C UBS Securities Australia Ltd Enter into Derivative C UBS Securities Australia Ltd Enter into Derivative C UBS Securities Australia Ltd Enter into Derivative C UBS Securities Australia Ltd Enter into Derivative C	MI-14	UBS Securities Australia Ltd	Enter into Derivative C	AN A	(3,000,000)	Options
UBS Securities Australia Ltd Enter into Derivative (UBS Securities Australia Ltd Enter into Derivative (UBS Securities Australia Ltd Enter into Derivative (UBS Securities Australia Ltd Enter into Derivative (UBS Securities Australia Ltd Enter into Derivative (Jun-14	UBS Securities Australia Ltd	Enter Into Derivative C	NA	(657)	Options
UBS Securities Australia Ltd Enter Into Derivative (UBS Securities Australia Ltd Enter Into Derivative (UBS Securities Australia Ltd Enter Into Derivative (UBS Securities Australia Ltd Enter Into Derivative (Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	(2,500)	Options
UBS Securities Australia Ltd Enter into Derivative (UBS Securities Australia Ltd Enter into Derivative (UBS Cacurities Australia Ltd Enter into Derivative (Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (ΝA	(3,500)	Options
UBS Securities Australia Ltd Enter Into Derivative C UBS Securities Australia Ltd Enter Into Derivative C	Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(279)	Options
UBS Securities Australia Ltd Enter Into Derivative Q	Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (ΑΝ	30	Options
	Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (ΑN	30	Options
UBS Securities Australia Ltd Foter Into Derivative C	lin-14	I IRS Securities Australia I td	Forter Into Derivative	NVA	(150)	,
Obsider Musical Ltd	1	Opp permittee Australia Fra	בנוננו בוני בעוואמוואי ל	. 17.77		0-4:00
ILIBS Societies Australia Ltd	:				(3/5)	Options

Buy Buy	8,109	836	Ordinary
Buy		333	Ordinary
	3.230	222	
2.5	31 903	3 289	Ordinan
Birv	30.264	3 120	Ordinary
Suy	340	35	Ordinary
Buy	7,120	734	Ordinary
Buy	40,342	4,159	Ordinary
Buy	23,270	2,399	Ordinary
Buy	89	7	Ordinary
Buv	20	٣	Ordinary
Bliv	4 375	ART	Ordinan
6.0	086.66	0000	Control of the contro
6	000	2	, and a
na a	67	^ 3	Ordinary
ng	066,1	107	Ordinary
Buy	(4)	//	Ordinary
Buy	1,562	191	Ordinary
Buy	2.192	226	Ordinary
Biv	10 032	1 1 2 7	Ordinary
200	6767	Cav	Conditions
, and	C#//#	409	Ordinary
Buy	29	3	Ordinary
Buy	4,918	205	Ordinary
Buv	20	~	Ordinan
, d	2	, ,,,	(1)
Social Control	765,67	3,020	Organary
Buy	7/7/78	3,32/	Ordinary
Buy	42,680	4,400	Ordinary
Buy	42.680	4.400	Ordinary
B	ONE	36	Ordinan
í c	000		0.00
had	976'9	114	Orginary
Buy	3,880	400	Ordinary
Buy	8,187	844	Ordinary
Ang	17.004	1.753	Ordinary
Riv	5 820	COO	Ordinary
600	000	3	i di di
Buy	67	3	Ordinary
Buy	29	m	Ordinary
Buy	33,999	3.505	Ordinary
, and	13 001	1001	Codings
(DA	100,01	160,1	or or or or
Buy	21,932	2,261	Ordinary
Buy	1,019	105	Ordinary
Buv	757	78	Ordinary
Bun	1 000	30,	Ordingo
(pg	920'1	200	Ordinary
BUŞ	01	-	Ordinary
Buy	14,550	1,500	Ordinary
Buv	2,192	326	Ordinary
Ben	C.	-	Crainon
600	2		Oldinary
Buy	22,310	2,300	Ordinary
Buy	15,181	1,565	Ordinary
a a	11 168	A 27E	Ordinan
(B		20,	100
boy	000,0	400	Ordinary
Buy	12,610	1,300	Ordinary
Buv	87	o	Ordinary
Bary	1 280	132	Ordinar,
60	007/	1000	(Sample)
had	19,400	2,000	Ordinary
Buy	37,112	3,826	Ordinary
BUV	34.047	3.510	Ordinary
Bus	VOC 0	000	Ordinan
, one	+07'0	ŧ60	Olollialy
Buy	23,765	2,450	Ordinary
Buy	41.574	4.286	Ordinary
2.8	15 510	1 500	Ordina
500		CCC,	in in in
Buy	3,880	400	Ordinary
Buy	7,227	745	Ordinary
S	35 066	3615	Ordinan
(50 d)	700,7	277	(mining)
and a	177'/	047	Countries
Buy	7#C,00	3,809	Ordinary
Buy	5,859	604	Ordinary
, a a	, , , , , , , , , , , , , , , , , , , ,	1	, , , , ,

26-Jun-14	Obs securities Australia Ltd	(60)			
6-1m-14	USS Securities Australia Ltd	pan)	3,734	38/	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buv	5.451	293	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buv	096	8	Ordinary
6-lin-14	BS Securities Australia and	Barx	1 775	66	Ordinary
26-hun-14	UBS Securities Australia I td	Bliv	2 512	250	Ordinary Ordinary
6-Jun-14	UBS Securities Australia Ltd	Bliv	2,5,5	G, ay	Ordinary
Jun-14	UBS Securities Australia Ltd	Buv	2.221	229	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	10	F	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	7,256	748	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	3,579	369	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	3,201	330	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	898'9	708	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	12,668	1.306	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	2,852	294	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	10,389	1,071	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Виу	29,410	3,032	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	1,096	113	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	2,959	305	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	2,648	273	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	437	45	Ordinary
- 1	UBS Securities Australia Ltd	Buy	757	78	Ordinary
- 1	UBS Securities Australia Ltd	Buy	1,562	161	Ordinary
- 1	UBS Securities Australia Ltd	Buy	7,071	729	Ordinary
	UBS Securities Australia Ltd	Buy	22,116	2,280	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	4,646	479	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	6,839	705	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	5,054	521	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	28,256	2,913	Ordinary
- In	UBS Securities Australia Ltd	Buy	1,135	711	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	892	95	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	1,588	174	Ordinary
20-Jun-14	UBS Securities Australia Ltd	Buy	6,412	196	Ordinary
20-Jun-14	UBS Securities Australia Ltd	Buy	3,182	328	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	2,561	264	Ordinary
un-14	UBS Securities Australia Ltd	Buy	4,229	436	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	8,507	877	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	6,858	707	Ordinary
- 1	UBS Securities Australia Ltd	Buy	8,943	922	Ordinary
- 1	UBS Securities Australia Ltd	Buy	12,086	1,246	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	5,122	528	Ordinary
- 1	UBS Securities Australia Ltd	Buy	2,309	238	Ordinary
	UBS Securities Australia Ltd	Buy	6,480	899	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	3,541	365	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	5,820	009	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	15,239	1,571	Ordinary
lun-14	UBS Securities Australia Ltd	Buy	3,104	320	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	31,040	3,200	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	10,117	1,043	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	6.421	662	Ordinary
	UBS Securities Australia Ltd	3nv	4.802	495	Ordinary
26-Jun-14	UBS Securities Australia I td	Buy	2 910	GG C	Ordinary
ı	UBS Securities Australia I td	Buy	24,213	2 564	Ordinary
ı	I IBS Securities Australia I to	200	0.53	100	Coding
26-Jun-14	UBS Securities Australia I td	Buy	8 643	200	Ordinary
un-14	UBS Securities Australia Ltd	Buv	1 154	110	Ordinacy
26-lun-14	I IRC Securities Australia 144	Bush	1067	100	Ordinary
± :	UBS Securities Australia Ltd	Buy	/90'1	011	Ordinary
20-JUN-14	UBS Securities Australia Ltd	Buy	20,748	2,139	Ordinary
-	UBS Securities Australia Ltd	Buy	10	-	Ordinary
- 1	UBS Securities Australia Ltd	Buy	22,068	2,275	Ordinary
- 1	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
26-lin-14	Late Commission Acceptable 144				
3	UBS Securities Austrana Liu	Buy	17,431	1,797	Ordinary

41.00	C. C. C. C. C. C. C. C. C. C. C. C. C. C		TAN.	ν. 10, –	Supriors
50-Jun-14	UBS Securities Australia Ltd	Correction Delivative of			
110	TIPO Contributo Australia Lid	Cotton Into Delivative	A A	7,00	Options
+1-10-02	ODS SECURICS Australia Liu	CHIEF IND DENVENE	₹.	,400	Options
100	ODS Securities Australia Ltd	Enter into Derivative (NA		Options
20-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (A/N	150	Options
26-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	177	Options
6-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	175	Options
Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (N/A	(170)	Options
26-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	(38)	Options
26-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (A/N	(220)	Options
56-Jun-14	UBS Securities Australia Ltd	Enter into Derivative d	ΨN	566	Options
26-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative d	AW	-	Ontions
6-lun-14	UBS Securities Australia I to	Forter Into Derivative (AVA	ç	Oritons
11	FIDS Countition Assets in 14d	Controlled Option	4774	2 6	2000
11 11 30	UDS SECURIES AUSTRAINS LIG	CITES INCOMENSATIVE	A	200	Options
-110r	Obs Global Asset Management	Seil	4,192	(431)	Ordinary
50-Jun-14	UBS Securities Australia Ltd	Buy	35,139	3,900	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	591,300	65,700	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Sell	096,550	(104,900)	Ordinary
Jun-14	UBS Securities Australia Ltd	les	962,500	(110,000)	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buv	3.237.500	350.000	Ordinary
6-Jun-14	UBS Securities Australia Ltd	35	200	(1001)	Ordinary
6-bin-14	I IBS Securities Australia I to	Birt	2 250 000	250,000	Ordinary
11 11	Libr Consistent Australia Lea	(ng	2,2,20,000	220,000	O'G'II BILD
	UBS Securities Australia LIG) Sei	9,9/4	(997,400)	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	212,500	25,000	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	328,125	37,500	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	258,075	27,900	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Sell	008	(100)	Ordinary
6-lin-14	LIRS Securities Australia Ltd	Riv	144 670	17.000	Ordinan
6-hrp-14	1 IBC Countries Australia 1+4	100	0,0,55	1000 6	diam's
1 1 1 1	COS Securities Australia Liu	Jac.	067,12	(000,6)	Ordinary
- Yau- 14	UBS Securities Australia Ltd	Buy	7,928	300	Ordinary
30n-14	UBS Securities Australia Ltd	Seil	3,542	(364)	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	35,046	3,613	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buv	8.381	864	Ordinary
6-lun-14	UBS Securities Australia (1d	Bro	2 832	292	Ordinary
14	I IBC Contribut Australia 144	0.00	2,00,2	707	Ordinal y
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1 DO Commission Australia Lat.	600	518	5 8	o dilliary
	Opp Securities Australia Ltd	mg	140	16	Cruriary
0-JULI-14	Obs securities Australia Ltd	Buy	1,75	183	Orginary
6-Jun-14	IUBS Securities Australia Ltd	Buy	4,511	465	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	35,948	3,706	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buv	24.134	2.488	Ordinary
6-11 p. 14	I BC Coourities Australia 14d	Bitte	2 040	300	Ordinan
-1100	Coo Securities Australia Liu	buy	2,949	304	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	6,877	709	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buy	9,331	362	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	008'9	701	Ordinary
6-Jun-14	UBS Securities Australia Ltd	185	131.085	(13,500)	Ordinary
26. lim. 14	I IBS Societies Australia Ltd	Bir	oc c	6) radioan
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	I IBC Committee Australia Ltd	600	67	2	Ordinary
1 1 1	Opp Securities Adstraila Little	, and	67	0	Ordinary
20-Jun-14	UBS Securities Australia Ltd	Buy	67	3.	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	2,910	300	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	3,337	344	Ordinary
6-Jun-14	UBS Securities Australia Ltd	Buv	19.837	2.045	Ordinary
6-lun-14	UBS Securities Australia Ltd	Buy	1 542	159	Ordinary
1 1 2	Laboration Assessment Dept.		2000		100
1000	Obs Securities Australia Ltd	hna	13,636	714')	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	485	20	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	2,910	300	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Biv	17.053	1 758	Ordinary
26-4 in-14	I IRS Securities Australia I to	Bry	13 202	1361	Ordinan
1 1	I no o a cultica Actualist tel	âng	302,01	100,1	Countries
26-Jun-14	UBS Securities Australia Ltd	Buy	5,296	246	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	6,567	677	Ordinary
26-Jun-14	UBS Securities Australia Ltd	Buv	920	1001	Ordinary
26-Jun-14	UBS Securities Australia 11d	Bix	2,619	022	Ordinary
6-1 m-14	I IBS Securities Australia I to	Bin	6 790	002	Ordinan
1.1.1.1	UDD Decutions Australia at the	ána	06/19	200/	Ordifiary
26-Jun-14	UBS Securities Australia Ltd	-			
i		pay	/1//1	177	Ordinary
IIII-14	UBS Securities Australia Ltd	Buv	14,249	1,469	Ordinary

30-Jun-14	ספס סביכתו נובס שמסתמוומ דנת	anà	20.100	200.00	Compary
30-Jun-14	I IRS Sacurities Australia Ltd	a	21 044	000.0	Carles O
30-1 m-14	UBS Securities Australia Ltd	Buy	8.043	830	Ordinary
-	UBS Securities Australia Ltd	Buy	2,265	234	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,149	222	Ordinary
10-Jun-14	UBS Securities Australia Ltd	Buy	0,670	666	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	20,163	2,083	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	20,948	2,164	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,168	224	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,410	249	Ordinary
0-Jun-14	i BC Countition Australia Ltd	puy P:10	2,120	240	Ordinary
30-lin-14	LIBS Securities Australia Ltd	Ser.	2,231	230	Ordinary
30-Jun-14	1085 Securities Australia Ltd	Berv	5,220	550	Ordinary
	UBS Securities Australia Ltd	Buy	300	31	Ordinary
	UBS Securities Australia Ltd	Buy	426	44	Ordinary
	UBS Securities Australia Ltd	Buy	1,646	170	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,526	261	Ordinary
- 1	UBS Securities Australia Ltd	Buy	2,410	249	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Buy	13,658	1,431	Ordinary
30-jun-14	UBS Securities Australia Ltd	Buy	21,160	2,186	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Buy	15,391	1,590	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	72,135	7,452	Ordinary
so-Jun-14	UBS Securities Australia Ltd	Buy	2,749	284	Ordinary
4 4	UBS Securities Australia Ltd	Buy	2,904	OOF S	Ordinary
30,100-14	LIRS Securities Australia 14d	and a	15004	1 550	Ordinary
30-lun-14	I IRS Securities Australia 11d	e de la companya de	4 240	OCC,1	Ordinary
30-lim-14	LIRS Securities Australia 11d	S. B.	907.0	1012	Ordinary
un-14	UBS Securities Australia Ltd	Buv	089'6	1 000	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	24,829	2,565	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,007	104	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	878	101	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	22,477	2,322	Ordinary
- 1	UBS Securities Australia Ltd	Buy	55,612	5,745	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	27,820	2,874	Ordinary
- 1	UBS Securities Australia Ltd	Buy	368	38	Ordinary
- 1	UBS Securities Australia Ltd	Buy	1,442	149	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	77,440	8,000	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	11,616	1,200	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	8,199	847	Ordinary
41-14	UBS Securities Australia Ltd	Buy	348	36	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,442	149	Ordinary
-14 -14	UBS Securities Australia Ltd	Buy	16,640	1,719	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	77,585	8,015	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	25,265	2,610	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,936	200	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	130,535	13,485	Ordinary
4 .	UBS Securities Australia Ltd	Buy	027'/7	718/7	Ordinary
30-Jun-14	UBS Securities Australia Ltd	BUJ	3,756	388	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	19,980	2,064	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	/,/44	800	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	14,549	1,503	Ordinary
un-14	UBS Securities Australia Ltd	Buy	3,272	338	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,759	285	Ordinary
41-I	UBS Securities Australia Ltd	Buy	755	27	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Buy	13,804	1,426	Ordinary
50-Jun-14	UDS Securities Australia Liu	ong.	76	2 8	Ordinary
30-Jun-14	UBS Securities Australia Ltd	gnà	899	69	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,381	246	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	13,881	1,434	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	13,155	1,359	Ordinary
-1	UBS Securities Australia Ltd	Buy	6,941	717	Ordinary
- 1	UBS Securities Australia Ltd	Buy	5,343	552	Ordinary
30-Jun-14	II IRS Securities Australia I to				

USS Securities Australia Ltd Buy 2.395 2.207	26-Jun-14	I IRS Securities Australia I td				,
Box Securities Autrafia Lut Byty 9,74 1,04 BOS Securities Autrafia Lut Byty 1,74 1,77 BOS Securities Autrafia Lut Byty 1,74 1,77 BOS Securities Autrafia Lut Byty 2,511 1,77 BOS Securities Autrafia Lut Byty 3,131 2,20 BOS Securities Autrafia Lut Byty 3,131 2,20 BOS Securities Autrafia Lut Byty 2,134 2,20 BOS Securities Autrafia Lut Byty 2,134 2,20 BOS Securities Autrafia Lut Byty 2,134 2,50 BOS Securities Autrafia Lut Byty 2,140 2,50 BOS Securities Autrafia Lut Byty 2,130 2,50 BOS Securities Autrafia Lut Byty 2,130 2,50 BOS Securities Autrafia Lut Byty	5 hin 11		Buy	2,395	247	Ordinary
Box 17,431 1,797 Box 17,431 1,797 Box 2,891 2,288 Box 2,891 2,288 Box 2,513 2,20 Box 2,513 2,23 Box 2,52 2,13 Box 2,17 3,10	0-JUL - 14	UBS Securities Australia Ltd	Buy	9,734	1,004	Ordinary
Big Securities Australia Ltd Bluy 17,715	6-Jun-14	UBS Securities Australia Ltd	Buy	10	1	Ordinary
Box Securities Australia Ltrd Buy 2,581 2,281 LOSS Securities Australia Ltrd Buy 3,511 1,771 LOSS Securities Australia Ltrd Buy 3,133 2,220 LOSS Securities Australia Ltrd Buy 2,732 2,877 LOSS Securities Australia Ltrd Buy 2,075 2,973 LOSS Securities Australia Ltrd Buy 1,052 2,973 LOSS Securities Australia Ltrd Buy 2,025 2,974 LOSS Securities Australia Ltrd Buy 2,025 2,974 LOSS Securitie	6-Jun-14	UBS Securities Australia Ltd	Buy	17,431	1,797	Ordinary
Box Securities Australie Ltd Buy 36,511 37,14 BS Securities Australie Ltd Buy 1,176 2,134 2,12 BS Securities Australie Ltd Buy 2,134 2,871 3,13 BS Securities Australie Ltd Buy 2,130 2,871 3,140 BS Securities Australie Ltd Buy 2,140 2,140 BS Securities Australie Ltd Buy 2,140 3,140 BS Securities Australie Ltd Buy 2,140 3,140 BS Securities Australie Ltd Buy 2,140 1,252 BS Securities Australie Ltd Buy 2,180 1,122 BS Securities Australie Ltd Buy 2,818 1,122 BS Securities Australie Ltd Buy 2,823 1,122 BS Securities Australie Ltd Buy 2,823 1,122 BS Securities Australie Ltd Buy 2,122 2,140 BS Securities Australie Ltd Buy 2,122 2,140 BS Securities Australie Ltd Buy 2,122 2,141	6-Jun-14	UBS Securities Australia Ltd	Buy	2,891	298	Ordinary
Boy 1,776 177 UBS Scentries Australie Ltd Buy 2,734 220 UBS Scentries Australie Ltd Buy 2,073 227 UBS Scentries Australie Ltd Buy 2,075 2,07 UBS Scentries Australie Ltd Buy 2,075 2,07 UBS Scentries Australie Ltd Buy 2,075 2,07 UBS Scentries Australie Ltd Buy 2,07 2,07 UBS Scentries Australie Ltd Buy 2,07 2,03 UBS Scentries Australie Ltd Buy 1,072,283 (1,083) UBS Scentries Australie Ltd Buy 1,072,283 (1,083) UBS Scentries Australie Ltd Buy 1,072,283 (1,083) UBS Scentries Australie Ltd Buy 1,052,083 (1,083) UBS Scentries Australie Ltd Buy 1,052,083 (1,083) UBS Scentries Australie Ltd Buy 1,052,083 (1,053) UBS Scentries Australie Ltd Buy 2,075 (4,430) UBS Scentries Australie Ltd Buy 2,075	6-Jun-14	UBS Securities Australia Ltd	Buy	36,511	3,764	Ordinary
UBS Securities Australia Ltd BM 3,134 22,00 2,817 UBS Securities Australia Ltd Buy 27,907 2,817 460 2,840 2,817 460 2,840 2,847 460 460 460 2,140 2,847 460	6-Jun-14	UBS Securities Australia Ltd	Buy	1,716	177	Ordinary
UBS Securities Australia Ltd Buy 3,133 228 UBS Securities Australia Ltd Buy 7,197 466 UBS Securities Australia Ltd Buy 7,071 467 471 UBS Securities Australia Ltd Buy 1,607 1,607 1,607 UBS Securities Australia Ltd Buy 1,607 1,607 1,728 UBS Securities Australia Ltd Buy 1,607 1,729 UBS Securities Australia Ltd Buy 1,262 1,729 UBS Securities Australia Ltd Buy 1,606 1,729 UBS Securities Australia Ltd Buy 2,005 1,729 UBS Securities Australia Ltd Buy 2,005 1,729	6-Jun-14	UBS Securities Australia Ltd	Buy	2,134	220	Ordinary
UBS Securities Australia Ltd Buy 27,907 2,877 UBS Securities Australia Ltd Buy 20,738 2,877 UBS Securities Australia Ltd Buy 20,738 2,140 UBS Securities Australia Ltd Buy 2,677 1,401 UBS Securities Australia Ltd Sell 1,507 (1,691) UBS Securities Australia Ltd Buy 2,878 (1,202) UBS Securities Australia Ltd Sell 1,502,839 (1,202) UBS Securities Australia Ltd Sell 1,602,839 (1,202) UBS Securities Australia Ltd Buy 2,973 (1,203) UBS Securities Australia Ltd Buy 2,973 (1,203) UBS Securities Australia Ltd Buy 2,073 (1,203) UBS Securities Australia Ltd Buy 2,073 (1,273) UBS Securities Australia Ltd Buy 2,073 (1,273) UBS Securities Australia Ltd Buy 2,073 (2,722) UBS Securities Australia Ltd Buy 2,072 (2,723) UBS Sec	6-Jun-14	UBS Securities Australia Ltd	Buy	3,133	323	Ordinary
UBS Securities Australia Ltd Buy A66 A171 UBS Securities Australia Ltd Buy A171 A171 UBS Securities Australia Ltd Buy A171 A171 UBS Securities Australia Ltd Buy A171 A172 UBS Securities Australia Ltd Buy A172 A172 UBS Securities Australia Ltd	5-Jun-14	UBS Securities Australia Ltd	Buy	706,72	2,877	Ordinary
UBS Securities Australia Ltd Buy 20,788 2,140 UBS Securities Australia Ltd Sell 4,071 (437) UBS Securities Australia Ltd Sell 1,677 (1437) UBS Securities Australia Ltd Buy 2,878 (1,207) UBS Securities Australia Ltd Sell 12,578 2,913 UBS Securities Australia Ltd Sell 12,578 (10,603) UBS Securities Australia Ltd Sell 10,606 (1,023) UBS Securities Australia Ltd Sell 10,606 (1,023) UBS Securities Australia Ltd Buy 8,81 10,606 UBS Securities Australia Ltd Buy 8,91 10,72 UBS Securities Australia Ltd Buy 8,91 10,43 UBS Securities Australia Ltd Buy 8,91 10,43 UBS Securities Australia	6-Jun-14	UBS Securities Australia Ltd	Buv	466	48	Ordinary
UBS Securities Australia Ltd Buy 4.171 (164) UBS Securities Australia Ltd Sell 1.2676 (1,301) UBS Securities Australia Ltd Buy 12,599 1,291 UBS Securities Australia Ltd Buy 12,599 1,291 UBS Securities Australia Ltd Buy 10,508 2,913 UBS Securities Australia Ltd Buy 10,600 2,913 UBS Securities Australia Ltd Buy 10,600 2,913 UBS Securities Australia Ltd Buy 10,600 1,739 UBS Securities Australia Ltd Buy 10,600 1,739 UBS Securities Australia Ltd Buy 2,15 2,15 UBS Securities Australia Ltd Buy 2,130 1,500 UBS Securities Australia Ltd Buy 2,130 1,500 UBS Securities Australia Ltd Buy 2,031 1,500 UBS Securities Australia Ltd Buy 2,032 1,570 UBS Securities Australia Ltd Buy 2,030 1,570 UBS Securities Australia Ltd	ı	UBS Securities Australia Ltd	Buv	20.758	2.140	Ordinary
UBS Securities Australia Ltd Sell 1,601 (164) UBS Securities Australia Ltd Buy 12,559 (1,370) UBS Securities Australia Ltd Buy 12,559 (1,572) UBS Securities Australia Ltd Sell Sell (8,782) (1,572) UBS Securities Australia Ltd Sell Sell (8,782) (8,782) (8,782) UBS Securities Australia Ltd Buy 1,662 (1,732) (1,732) (1,732) UBS Securities Australia Ltd Buy 1,662 (1,732) (1,732) (1,732) UBS Securities Australia Ltd Buy 5,913 (1,732) (1,732) (1,732) UBS Securities Australia Ltd Buy 5,913 (1,732) (1,732) (1,732) UBS Securities Australia Ltd Buy 5,913 (1,732) (1,732) (1,732) (1,732) UBS Securities Australia Ltd Buy 5,913 (1,732) (1,732) (1,732) (1,732) (1,732) (1,732) (1,732) (1,732) (1,732) (1,732)	•	UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
UBS Securities Australia Ltd Skill 112,576 (1,301) UBS Securities Australia Ltd Buy 12,579 2,913 UBS Securities Australia Ltd Skill 10,52,885 (1,82,391) UBS Securities Australia Ltd Skill 10,52,885 (1,82,392) UBS Securities Australia Ltd Skill 10,660 (1,093) UBS Securities Australia Ltd Buy 2,15 2,24 UBS Securities Australia Ltd Buy 8,21 2,15 UBS Securities Australia Ltd Buy 8,91 2,15 UBS Securities Australia Ltd Buy 8,91 3,00 UBS Securities Australia Ltd Buy 8,91 4,90 UBS Securities Australia Ltd Buy 2,023 1,59 UBS Securities Australia Ltd Buy 4,90 1,50 UBS Securities Australia Ltd Buy 4,90 5,001 UBS Securities Australia Ltd Buy 4,90 5,001 UBS Securities Australia Ltd Buy 4,90 5,001 UBS Securities Austrialia	ı	UBS Securities Australia Ltd	J.S.	1,601	(164)	Ordinary
UBS Securities Australia Ltd Buy 2,813 2,913 UBS Securities Australia Ltd Stall 1,052,895 1,622,895 UBS Securities Australia Ltd Stall 1,052,895 1,052,895 UBS Securities Australia Ltd Buy 1,052,805 1,052,006 UBS Securities Australia Ltd Buy 2,915 1,052,008 UBS Securities Australia Ltd Buy 2,917 2,917 UBS Securities Australia Ltd Buy 2,917 3,006 UBS Securities Australia Ltd Buy 2,917 3,007 UBS Securities Australia Ltd Buy 2,917 3,007 UBS Securities Australia Ltd Buy 2,917 4,43,00 UBS Securities Australia Ltd Buy 2,917 4,43,00 UBS Securities Australia Ltd Buy 2,906 1,02,00 UBS Securities Australia Ltd Buy 2,906 1,02,00 UBS Securities Australia Ltd Buy 2,906 1,02,00 UBS Securities Australia Ltd Buy 2,906 2,00 UB		UBS Securities Australia Ltd	Sell	12,676		Ordinary
UBS Securities Australia Ltd Sell 1052,899 (1,879) UBS Securities Australia Ltd Sell 105,289 (1,879) UBS Securities Australia Ltd Sell 16,560 (1,093) UBS Securities Australia Ltd Buy 10,660 (1,093) UBS Securities Australia Ltd Buy 275 22 UBS Securities Australia Ltd Buy 275 22 UBS Securities Australia Ltd Buy 275 22 UBS Securities Australia Ltd Buy 20,11,27 22 UBS Securities Australia Ltd Buy 20,11,27 14,430 UBS Securities Australia Ltd Buy 20,288 15,709 UBS Securities Australia Ltd Buy 20,288 14,430 UBS Securities Australia Ltd Buy 20,288 14,430 UBS Securities Australia Ltd Buy 20,288 14,430 UBS Securities Australia Ltd Buy 20,288 1,001 UBS Securities Australia Ltd Buy 20,208 1,001 UBS Securities Australia L	7-Jun-14	UBS Securities Australia Ltd	Buy	28,330		Ordinary
UBS Securities Australia Ltd Sell 1,052,885 (1,08,039) UBS Securities Australia Ltd Buy 1,6,661 (1,029) UBS Securities Australia Ltd Sell 10,6,601 (1,029) UBS Securities Australia Ltd Buy 23,555 22 UBS Securities Australia Ltd Buy 23,555 22 UBS Securities Australia Ltd Buy 23,555 22 UBS Securities Australia Ltd Buy 20,503 84 UBS Securities Australia Ltd Buy 20,503 84 UBS Act London Branch Sell 430,051 (4,4,30) UBS Securities Australia Ltd Buy 20,528 (2,03) UBS Securities Australia Ltd Buy 430,051 (4,4,30) UBS Securities Australia Ltd Buy 20,528 (2,03) UBS Securities Australia Ltd Buy 20,528 (2,03) UBS Securities Australia Ltd Buy 20,528 (2,438) UBS Securities Australia Ltd Buy 20,528 (2,438) UBS Securities Au	7-Jun-14	UBS Securities Australia Ltd	Buy	12,599		Ordinary
UBS Securities Australia Ltd Sell In 146,622 (6,793) UBS Securities Australia Ltd Sini 10,660 15,044 3,006 UBS Securities Australia Ltd Sell 10,660 (1,033) 10,660 (1,033) UBS Securities Australia Ltd Buy 215 22 22 UBS Securities Australia Ltd Buy 811,305 84,127 UBS Securities Australia Ltd Sell 430,051 (44,330) UBS Securities Australia Ltd Sell 430,051 (44,301) UBS Securities Australia Ltd Buy 86,834 86,550 UBS Securities Australia Ltd Buy 48,007 5,091 UBS Securities Australia Ltd Buy 48,007 5,091 UBS Securities Australia Ltd Buy 11,443 11,443 UBS Securities Australia Ltd <	7-Jun-14	UBS Securities Australia Ltd	les.	1,052,835		Ordinary
UBS Securities Australia Ltd Bluy 146,628 15,044 UBS Securities Australia Ltd Sell 1,056 (1,739) UBS Vecurities Australia Ltd Sell 1,056 (1,739) UBS Vecurities Australia Ltd Buy 59,38 6,083 UBS Securities Australia Ltd Buy 215 6,22 UBS Securities Australia Ltd Buy 911,121 (1,739) UBS Securities Australia Ltd Buy 911,121 (4,430) UBS Securities Australia Ltd Buy 911,121 (6,071) UBS Securities Australia Ltd Buy 911,121 (6,071) UBS Securities Australia Ltd Buy 911,121 (6,071) UBS Securities Australia Ltd Buy 430,032 (6,071) UBS Securities Australia Ltd Buy 430,032 (6,071) UBS Securities Australia Ltd Buy 44,000 (6,071) UBS Securities Australia Ltd Buy 46,000 (6,071) UBS Securities Australia Ltd Buy 46,000 (6,071) <td< td=""><td>7-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>688'58</td><td></td><td>Ordinary</td></td<>	7-Jun-14	UBS Securities Australia Ltd	Sell	688'58		Ordinary
URS Securities Australia Ltd Entil Derivative 10,660 (1,039) UBS Securities Australia Ltd Sell 10,660 (1,039) UBS Wealth Management Australia Ltd Buy 53,88 6,088 UBS Securities Australia Ltd Buy 20,130 84 UBS Securities Australia Ltd Buy 911,305 94,127 UBS Securities Australia Ltd Buy 911,305 94,127 UBS Securities Australia Ltd Sell 911,305 94,127 UBS Securities Australia Ltd Sell 94,637 (5,091) UBS Securities Australia Ltd Sell 430,03 (5,091) UBS Securities Australia Ltd Buy 20,58 (5,091) UBS Securities Australia Ltd Buy 20,58 (5,091) UBS Securities Australia Ltd Buy 20,58 (7,39) UBS Securities Australia Ltd Buy 20,58 (2,438) UBS Securities Australia Ltd Buy 20,58 (2,438) UBS Securities Australia Ltd Buy 43,000 (2,438)	7-jun-14	UBS Securities Australia Ltd	Buy	146,628		Ordinary
UBS Wealth Management Australia Ltd Sell 10,660 (1,039) UBS Wealth Management Australia Ltd 941 59,388 6,088 UBS Securities Australia Ltd Buy 59,388 6,088 UBS Securities Australia Ltd Buy 820 64 UBS Securities Australia Ltd Buy 911,305 94,127 UBS Securities Australia Ltd Buy 911,305 94,127 UBS ACC London Banch Buy 911,307 94,127 UBS ACC London Banch Buy 91,436 94,127 UBS ACC London Banch Buy 91,436 94,127 UBS Securities Australia Ltd Buy 91,436 94,127 UBS Securities Australia Ltd Buy 90,73 10,071 UBS Securities Australia Ltd Buy 20,258 20,01 UBS Securities Australia Ltd Buy 48,000 1,44,30) UBS Securities Australia Ltd Buy 48,000 1,44,30) UBS Securities Australia Ltd Buy 48,000 1,1,31 UBS Securities Australia	7-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (ΝA		Options
UBS Securities Australia Ltd Sell 16,981 (1,739) UBS Securities Australia Ltd Buy 59,338 6,088 UBS Securities Australia Ltd Buy 827 82 UBS Securities Australia Ltd Buy 81,130 84,127 UBS AG London Branch Sell 91,1305 14,430 UBS AG London Branch Sell 91,1305 14,430 UBS AG London Branch Sell 91,1305 14,430 UBS AG London Branch Sell 91,131 14,430 UBS Securities Australia Ltd Buy 15,432 16,730 UBS Securities Australia Ltd Buy 20,258 2,091 UBS Securities Australia Ltd Buy 48,005 1,144 1,144 UBS Securities Australia Ltd Buy </td <td>7-Jun-14</td> <td>UBS Wealth Management Australia Ltd</td> <td>Seil</td> <td>10,660</td> <td>(1,093)</td> <td>Ordinary</td>	7-Jun-14	UBS Wealth Management Australia Ltd	Seil	10,660	(1,093)	Ordinary
UBS Securities Australia Ltd Buy 59,358 6,088 UBS Securities Australia Ltd Buy 215 22 UBS Securities Australia Ltd Buy 911,301 94,127 UBS Securities Australia Ltd Buy 911,301 94,127 UBS Accurities Australia Ltd Buy 91,131 94,130 UBS Securities Australia Ltd Buy 430,051 44,300 UBS Securities Australia Ltd Buy 9,073 (5,071) UBS Securities Australia Ltd Sell 49,073 (5,071) UBS Securities Australia Ltd Buy 49,073 (5,071) UBS Securities Australia Ltd Buy 48,000 21,370 UBS Securities Australia Ltd Buy 48,000 21,370 UBS Securities Australia Ltd Buy 48,000 21,370 UBS Securities Australia Ltd Buy 48,000 21,380 UBS Securities Australia Ltd Buy 48,700 1,184 UBS Securities Australia Ltd Buy 48,700 1,184 UBS Securities Au	7-hin-14	UBS Wealth Management Australia Ltd	Sell	196'91	(1.739)	Ordinary
UBS Securities Australia Ltd Buy 215 22 UBS Securities Australia Ltd Buy 911,302 94,127 UBS AG London Branch Seli 911,201 64,127 UBS AG London Branch Buy 911,301 64,127 UBS AG London Branch Buy 91,301 (44,30) UBS AG London Branch Buy 490,031 (44,30) UBS Securities Australia Ltd Buy 20,28 (47,30) UBS Securities Australia Ltd Buy 20,28 (5,71) UBS Securities Australia Ltd Buy 48,400 (5,00) UBS Securities Australia Ltd Buy 20,600 (5,00) UBS Securities Australia Ltd Buy 46,007 (7,00) UBS Securities Australia Ltd Buy 45,007 (4,430) UBS Securities Australia Ltd Buy	7-in-14	UBS Securities Australia Ltd	Buv	59.358	6.088	Ordinary
UBS Securities Australia Ltd Buy B20 B41 UBS Securities Australia Ltd Buy 91,305 94,127 UBS AG London Branch Sell 911,121 (94,127) UBS AG London Branch Buy 91,11,121 (94,127) UBS AG London Branch Buy 430,142 44,300 UBS AG London Branch Buy 15,147 (977) UBS Securities Australia Ltd Buy 15,147 (977) UBS Securities Australia Ltd Buy 20,258 2,071 UBS Securities Australia Ltd Buy 48,005 1,054 UBS Securities Australia Ltd Buy 88,888 2,000 UBS Securities Australia Ltd Buy 88,888 2,000 UBS Securities Australia Ltd Buy 88,888 2,000 UBS Securities Australia Ltd Buy 86,927 1,184 UBS Securities Australia Ltd Buy 86,927 3,271 UBS Securities Australia Ltd Buy 86,927 3,281 UBS Securities Australia Ltd Buy </td <td>7-810-14</td> <td>LIBS Securities Australia I td</td> <td>Buy</td> <td>215</td> <td>22</td> <td>Ordinary</td>	7-810-14	LIBS Securities Australia I td	Buy	215	22	Ordinary
UBS AG London Branch Buy 911,305 94,127 UBS AG London Branch Sell 911,305 (44,30) UBS AG London Branch Sell 430,142 (44,30) UBS AG London Branch Buy 430,142 (44,30) UBS AG London Branch Buy 430,142 44,430 UBS AG London Branch Buy 430,142 44,430 UBS Securities Australia Ltd Sell 430,123 (507) UBS Securities Australia Ltd Buy 430,051 (44,30) UBS Securities Australia Ltd Buy 430,051 (44,30) UBS Securities Australia Ltd Buy 480,000 (5,07) UBS Securities Australia Ltd Buy 480,000 (2,438) UBS Securities Australia Ltd	7-lin-14	UBS Securities Australia Ltd	Buv	820	84	Ordinary
UBS AG London Branch Sell 911,121 (44,30) UBS AG London Branch Sell 94051 44,430 UBS AG London Branch Buy 430,142 44,430 UBS Securities Australia Ltd Buy 430,137 44,430 UBS Securities Australia Ltd Buy 20,258 2,091 UBS Securities Australia Ltd Buy 20,598 2,091 UBS Securities Australia Ltd Buy 48,000 1,564 UBS Securities Australia Ltd Buy 43,762 5,000 UBS Securities Australia Ltd Buy 43,762 5,148 UBS Securities Australia Ltd Buy	0-hip-14	I IBS AG London Branch	Birv	911 305	94 127	Ordinary
UBS AGE (notion Parach) Sell 430,142 (44,430) UBS AG (condon Branch) Buy 430,142 44,430 UBS AG (condon Branch) Buy 430,142 44,430 UBS Securities Australia Ltd Sell 49,073 (5,071) UBS Securities Australia Ltd Buy 20,258 2,091 UBS Securities Australia Ltd Sell 430,051 (44,430) UBS Securities Australia Ltd Buy 430,051 (44,430) UBS Securities Australia Ltd Buy 48,000 5,000 UBS Securities Australia Ltd Buy 48,000 1,144 UBS Securities Australia Ltd Buy 48,000 1,144 UBS Securities Australia	1000	John Control States	Coll	011 121	121,120	Ordinary
UBS Securities Australia Ltd Buy 430,145 (44,430) UBS AG London Name Buy 430,145 15,660 UBS Securities Australia Ltd Buy 151,476 15,600 UBS Securities Australia Ltd Buy 20,238 2,091 UBS Securities Australia Ltd Buy 20,238 2,091 UBS Securities Australia Ltd Sell 43,005 1,5798 UBS Securities Australia Ltd Buy 20,238 2,091 UBS Securities Australia Ltd Buy 20,693 2,127 UBS Securities Australia Ltd Buy 20,693 1,127 UBS Securities Australia Ltd Buy 20,693 1,145 UBS Securities Australia Ltd Buy 450,073 1,145 UBS Securities Australia Ltd Buy 450,073 1,144 UBS Securities Australi	11-101-0	CBS AG LOUNGII DISHIGI	17-1	121,116	(121,45)	Oldinary Oldinary
UBS Securities Australia Ind Buy 44,30 UBS Securities Australia Ind 8el 9,463 (977) UBS Securities Australia Ind 8el 9,463 (977) UBS Securities Australia Ind 8el 30,73 (5,071) UBS Securities Australia Ind 8el 31,121 (44,430) UBS Securities Australia Ind 8uy 20,580 2,000 UBS Securities Australia Ind 8uy 20,590 5,000 UBS Securities Australia Ind 8uy 20,590 5,000 UBS Securities Australia Ind 8uy 868,834 8,137 UBS Securities Australia Ind 8uy 130,129 1,554 UBS Securities Australia Ind 8uy 49,762 8,146 UBS Securities Australia Ind 8uy 49,762 1,184 UBS Securities Australia Ind 8uy 49,762 1,184 UBS Securities Australia Ind 8uy 86,927 3,567 UBS Securities Australia Ind 8uy 36,000 1,231 UBS Securities Australia Ind 8u	0-Jun-14	UBS AG London Branch	Sell	430,051	(44,430)	Orginary
UBS Securities Australia Ltd Buy 15,600 UBS Securities Australia Ltd Sell 9,463 (5,071) UBS Securities Australia Ltd Sell 49,073 (5,071) UBS Securities Australia Ltd Sell 430,053 (44,420) UBS Securities Australia Ltd Buy 20,690 2,798 UBS Securities Australia Ltd Buy 48,688,34 89,687 UBS Securities Australia Ltd Buy 20,690 2,1370 UBS Securities Australia Ltd Buy 20,690 2,1370 UBS Securities Australia Ltd Buy 49,762 2,148 UBS Securities Australia Ltd Buy 49,762 2,146 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 80,985 5,567 UBS Securities Australia Ltd Buy 80,987 5,508 6,179 UBS Securiti	0-Jun-14	UBS AG London Branch	Buy	430,142	44,430	Ordinary
UBS Securities Australia Ltd Sell 49.63 (977) UBS Securities Australia Ltd Sell 49.073 (5.071) UBS Securities Australia Ltd Sell 20,258 2,091 UBS Securities Australia Ltd Sell 430,051 (44,430) UBS Securities Australia Ltd Buy 20,599 (75,798) UBS Securities Australia Ltd Buy 20,899 75,798 UBS Securities Australia Ltd Buy 20,898 83,400 5,000 UBS Securities Australia Ltd Buy 45,007 1,654 1,654 UBS Securities Australia Ltd Buy 45,007 1,448 1,448 UBS Securities Australia Ltd Buy 45,007 1,448 1,144 UBS Securities Australia Ltd Buy 43,762 5,799 1,144 UBS Securities Australia Ltd Buy 43,762 5,799 1,144 UBS Securities Australia Ltd Buy 86,924 8,799 1,144 UBS Securities Australia Ltd Buy 88,900 1,149 1,149	0-Jun-14	UBS Securities Australia Ltd	Buy	151,476	15,660	Ordinary
UBS Securities Australia Ltd Sell 49,073 (5,071) UBS Securities Australia Ltd Buy 20,258 2,091 UBS Securities Australia Ltd Sell 430,051 (44,430) UBS Securities Australia Ltd Buy 48,005 5,000 UBS Securities Australia Ltd Buy 48,400 5,000 UBS Securities Australia Ltd Buy 48,400 5,1370 UBS Securities Australia Ltd Buy 48,000 2,1370 UBS Securities Australia Ltd Buy 48,000 2,1370 UBS Securities Australia Ltd Buy 48,007 2,488 UBS Securities Australia Ltd Buy 48,007 1,448 UBS Securities Australia Ltd Buy 48,007 4,430 UBS Securities Australia Ltd Buy 48,007 4,430 UBS Securities Australia Ltd Buy 4,900 1,144 UBS Securities Australia Ltd Buy 5,908 6,17 UBS Securities Australia Ltd Buy 5,908 6,17 UBS Securities Austral	0-Jun-14	UBS Securities Australia Ltd	Sell	9,463	(476)	Ordinary
UBS Securities Australia Ltd Buy 20,258 2,091 UBS Securities Australia Ltd Sell 44,400 (44,420) UBS Securities Australia Ltd Buy 26,900 7,5798) UBS Securities Australia Ltd Buy 26,900 2,300 UBS Securities Australia Ltd Buy 26,834 89,687 UBS Securities Australia Ltd Buy 20,699 21,370 UBS Securities Australia Ltd Buy 20,690 21,370 UBS Securities Australia Ltd Buy 450,071 46,533 UBS Securities Australia Ltd Buy 450,072 46,533 UBS Securities Australia Ltd Buy 450,071 46,533 UBS Securities Australia Ltd Buy 450,071 46,533 UBS Securities Australia Ltd Buy 450,071 46,573 UBS Securities Australia Ltd Buy 86,086 5,184 UBS Securities Australia Ltd Buy 86,086 5,184 UBS Securities Australia Ltd Buy 86,086 5,184 UBS Secur	0-Jun-14	UBS Securities Australia Ltd	Sell	£20'67	(5,071)	Ordinary
UBS Securities Australia Ltd Sell 430,051 (44,430) UBS Securities Australia Ltd Sell 91,1121 (64,127) UBS Securities Australia Ltd Buy 26,590 5,000 UBS Securities Australia Ltd Buy 26,590 5,000 UBS Securities Australia Ltd Buy 26,590 5,1370 UBS Securities Australia Ltd Buy 26,590 2,1370 UBS Securities Australia Ltd Buy 23,600 (2,438) UBS Securities Australia Ltd Buy 49,722 5,146 UBS Securities Australia Ltd Buy 49,722 5,146 UBS Securities Australia Ltd Buy 49,722 5,146 UBS Securities Australia Ltd Buy 86,927 1,184 UBS Securities Australia Ltd Buy 86,927 3,567 UBS Securities Australia Ltd Buy 5,483 5,52 UBS Securities Australia Ltd Buy 5,483 5,52 UBS Securities Australia Ltd Buy 5,433 6,57 UBS Securities Austr	0-Jun-14	UBS Securities Australia Ltd	Buy	20,258	2,091	Ordinary
UBS Securities Australia Ltd Sell 911,121 (94,127) UBS Securities Australia Ltd Buy 48,400 5,000 UBS Securities Australia Ltd Buy 206,990 21,370 UBS Securities Australia Ltd Buy 206,990 21,370 UBS Securities Australia Ltd Enter Into Derivative C NAA 1,684 UBS Securities Australia Ltd Buy 49,722 13,457 UBS Securities Australia Ltd Buy 49,722 5,146 UBS Securities Australia Ltd Buy 49,722 5,146 UBS Securities Australia Ltd Buy 49,722 5,146 UBS Securities Australia Ltd Buy 85,086 5,739 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 86,224 8,787 UBS Securities Australia Ltd Buy 8,437 3,767 UBS Securities Australia Ltd Buy 8,230 6,437 UBS	0-lun-14	UBS Securities Australia Ltd	Sell	430.051	(44,430)	Ordinary
UBS Securities Australia Ltd Sell 733,329 (75,798) UBS Securities Australia Ltd Buy 48,400 5,000 UBS Securities Australia Ltd Buy 206,990 21,300 UBS Securities Australia Ltd Buy 868,834 89,683 UBS Securities Australia Ltd Buy 43,600 1,564 UBS Securities Australia Ltd Buy 43,762 5,146 UBS Securities Australia Ltd Buy 43,762 5,146 UBS Securities Australia Ltd Buy 43,762 5,146 UBS Securities Australia Ltd Buy 86,087 8,799 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 86,924 8,506 UBS Securities Australia Ltd Buy 86,924 8,506 UBS Securities Australia Ltd Buy 5,908 5,508 UBS Securities Austr	14	I IBC Sequenties Australia 14d	Soll	911121	(701 107)	Ordinary
UBS Securities Australia Ltd Buy 48,400 V. J.729 UBS Securities Australia Ltd Buy 26,990 21,370 UBS Securities Australia Ltd Buy 26,990 21,370 UBS Securities Australia Ltd Enter Into Derivative C NA 2,868 UBS Securities Australia Ltd Buy 130,120 13,489 UBS Securities Australia Ltd Buy 450,071 46,543 UBS Securities Australia Ltd Buy 5,483 8,989 UBS Securities Australia Ltd Buy 5,483 8,989 UBS Securities Australia Ltd Buy 5,908 6,71 UBS Securities Australia Ltd Buy 5,908 6,71 UBS Securities Australia Ltd Buy 5,908 6,71 UBS	14	100 Contribute Australia I tol	Coll	000 557		Ordinary
UBS Securities Australia Ltd Buy 48,400 1,500 UBS Securities Australia Ltd Buy 206,900 21,300 UBS Securities Australia Ltd Buy 868,834 89,687 UBS Securities Australia Ltd Enter Into Derivative NA 2,684 UBS Securities Australia Ltd Buy 1,654 1,654 UBS Securities Australia Ltd Buy 450,071 4,653 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 86,786 8,799 UBS Securities Australia Ltd Buy 86,786 6,773 UBS Securities Australia Ltd Buy 86,724 8,986 UBS Securities Australia Ltd Buy 8,780 6,677 UBS Securities Australia Ltd Buy 5,908 6,114 UBS Securities Australia Ltd Buy 5,908 6,17 UBS Securities	0-Jun-14	UDS Securities Australia Ltd	Jac. 1	626,667		Cluriary
UBS Securities Australia Ltd Buy 20,590 21,370 UBS Securities Australia Ltd Enter into Derivative NA 1,654 UBS Securities Australia Ltd Enter into Derivative NA 1,654 UBS Securities Australia Ltd Buy 130,129 13,457 UBS Securities Australia Ltd Buy 49,762 1,684 UBS Securities Australia Ltd Buy 49,762 1,684 UBS Securities Australia Ltd Buy 86,924 8,799 UBS Securities Australia Ltd Buy 86,924 8,799 UBS Securities Australia Ltd Buy 86,924 8,99 UBS Securities Australia Ltd Buy 86,924 8,92 UBS Securities Australia Ltd Buy 86,924 8,50 UBS Securities Australia Ltd Buy 3,642 3,567 UBS Securities Australia Ltd Buy 3,642 3,567 UBS Securities Australia Ltd Buy 5,908 6,457 UBS Securities Australia Ltd Buy 5,908 6,457 UBS	0-Jun-14	UBS Securities Australia Ltd	Buy	48,400		Orginary
UBS Securities Australia Ltd Buy BBS 834 89 687 UBS Securities Australia Ltd Enter Into Derivative NA 1,684 UBS Securities Australia Ltd Buy 23,600 (2,438) UBS Securities Australia Ltd Buy 49,727 46,543 UBS Securities Australia Ltd Buy 49,727 5,146 UBS Securities Australia Ltd Buy 49,727 5,146 UBS Securities Australia Ltd Buy 49,727 5,63 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 8,427 3,767 UBS Securities Australia Ltd Buy 5,908 6,611 UBS Securities Australia Ltd Buy 5,908 6,611 UBS Securities Australia Ltd Buy 5,908 6,617 UBS Securities	0-Jun-14	UBS Securities Australia Ltd	Buy	206,990		Ordinary
UBS Securities Australia Ltd Finer Into Derivative NAA 1,654 UBS Securities Australia Ltd Sell 2,868 UBS Securities Australia Ltd Buy 45,0071 46,543 UBS Securities Australia Ltd Buy 43,762 5,146 UBS Securities Australia Ltd Buy 43,762 5,146 UBS Securities Australia Ltd Buy 86,904 1,144 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 3,671 1,134 UBS Securities Australia Ltd Buy 36,727 3,767 UBS Securities Australia Ltd Buy 36,427 3,767 UBS Securities Australia Ltd Buy 5,508 6,11 UBS Securities Australia Ltd Buy 5,508 6,457 UBS Securities Australia Ltd Buy 5,508 6,457 UBS Securities Australia Ltd Buy 5,508 6,439 16 UBS Securities Austr	0-Jun-14	UBS Securities Australia Ltd	8uy	868,834		Ordinary
UBS Securities Australia Ltd Enter Into Derivative (NVA 2.888 UBS Securities Australia Ltd Buy 130,129 13,459 UBS Securities Australia Ltd Buy 450,071 46,543 UBS Securities Australia Ltd Buy 450,071 46,543 UBS Securities Australia Ltd Buy 85,086 8,799 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 5,483 567 UBS Securities Australia Ltd Buy 5,483 3,961 UBS Securities Australia Ltd Buy 5,908 6,11 UBS Securities Australia Ltd Buy 5,908 6,61 UBS Securities Australia Ltd Buy 5,908 6,61 UBS Securities Australia Ltd Buy 5,780 6,66 UBS Securities Australia Ltd Buy 5,780 6,66 UBS Securities Australia Ltd Buy 5,308 6,61 UBS Securities Australia Ltd Buy 5,308 6,67 UBS Securities Australi	0-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (Ψ.N		Options
UBS Securities Australia Ltd Sell 23,600 (2,438) UBS Securities Australia Ltd Buy 450,721 13,457 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 49,762 5,146 UBS Securities Australia Ltd Buy 46,294 1,144 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 86,204 1,164 UBS Securities Australia Ltd Buy 36,727 1,57 UBS Securities Australia Ltd Buy 36,208 6,11 UBS Securities Australia Ltd Buy 5,908 6,17 UBS Securities Australia Ltd	0-Jun-14	UBS Securities Australia Ltd	Enter Into Derivative (Ą		Options
UBS Securities Australia Ltd Buy 130,129 13,457 UBS Securities Australia Ltd Buy 49,027 46,543 UBS Securities Australia Ltd Buy 49,027 5,146 UBS Securities Australia Ltd Buy 85,086 8,799 UBS Securities Australia Ltd Buy 5,883 5,787 UBS Securities Australia Ltd Buy 5,883 5,527 UBS Securities Australia Ltd Buy 38,303 3,961 UBS Securities Australia Ltd Buy 3,6427 3,767 UBS Securities Australia Ltd Buy 5,908 611 UBS Securities Australia Ltd Buy 5,908 613 UBS Securities Australia Ltd Buy 5,908 6437 UBS Securities Australia Ltd Buy 5,308 16 UBS Securities Australia Ltd <td< td=""><td>0-lin-14</td><td>I IBS Wealth Management Australia Ltd</td><td>lley</td><td>23,600</td><td>(2.438)</td><td>Ordinary</td></td<>	0-lin-14	I IBS Wealth Management Australia Ltd	lley	23,600	(2.438)	Ordinary
UBS Securities Australia Ltd Buy 450,071 46,543 UBS Securities Australia Ltd Buy 43,762 5,146 UBS Securities Australia Ltd Buy 85,086 8,799 UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 8,692 8,989 UBS Securities Australia Ltd Buy 38,303 3,961 UBS Securities Australia Ltd Buy 38,427 3,767 UBS Securities Australia Ltd Buy 5,908 6,611 UBS Securities Australia Ltd Buy 5,508 6,677 UBS Securities Australia Ltd Buy 5,308 6,677 UBS Securities Australia Ltd Buy 5,308 6,677 UBS Securities Australia Ltd Buy 62,473 5,506 UBS Securities Australia Ltd Buy 5,343 5,506 UBS Securities Australia Ltd Buy 5,343 5,506 UBS Securities Australia Ltd Buy 259 30 UBS Securities Australia Ltd	0-lun-14	LIBS Securities Australia Itd	Buy	130 129	13 457	Ordinary
UBS Securities Australia Ltd Buy 43,762 43,782	11.10	Let all report to the last of	, n	AEO 021	CESTON	Ordinan
UBS Securities Australia Ltd Buy 44,762 1,149 1,149 1,149 1,149 1,149 1,149 1,184	0-JUN-14	ODS Securities Australia Ltd	an)	170,064	0+0,0+2	Comany
UBS Securities Australia Ltd Buy 86,798 8,799 UBS Securities Australia Ltd Buy 11,449 1,144 1,144 UBS Securities Australia Ltd Buy 86,924 8,989 8,989 UBS Securities Australia Ltd Buy 11,904 1,231 1,271 UBS Securities Australia Ltd Buy 38,303 3,961 1,087 1,087 UBS Securities Australia Ltd Buy 5,908 6,11 1,087 2,666 UBS Securities Australia Ltd Buy 5,908 6,11 1,087 2,666 UBS Securities Australia Ltd Buy 6,2439 6,427 1,66 UBS Securities Australia Ltd Buy 5,308 6,477 1,66 UBS Securities Australia Ltd Buy 5,308 6,477 1,66 UBS Securities Australia Ltd Buy 33,243 5,506 1,66 UBS Securities Australia Ltd Buy 33,243 5,506 1,66 UBS Securities Australia Ltd Buy 2,30 30 1,66 <td>0-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>49,762</td> <td>5,146</td> <td>Ordinary</td>	0-Jun-14	UBS Securities Australia Ltd	Buy	49,762	5,146	Ordinary
UBS Securities Australia Ltd Buy 11.449 1.184 UBS Securities Australia Ltd Buy 6.924 6.989 UBS Securities Australia Ltd Buy 5.483 5.67 UBS Securities Australia Ltd Buy 11.904 1.231 UBS Securities Australia Ltd Buy 38,303 3.961 UBS Securities Australia Ltd Buy 35,088 5,522 UBS Securities Australia Ltd Buy 5,780 641 UBS Securities Australia Ltd Buy 52,780 2,666 UBS Securities Australia Ltd Buy 52,433 5,506 UBS Securities Australia Ltd Buy 52,780 16 UBS Securities Australia Ltd Buy 82,433 5,506 UBS Securities Australia Ltd Buy 83,243 5,506 UBS Securities Australia Ltd Buy 83,243 35 UBS Securities Australia Ltd Buy 83,243 35 UBS Securities Australia Ltd Buy 83,243 35 UBS Securities Australia Ltd	0-Jun-14	UBS Securities Australia Ltd	Buy	82,086	8,799	Ordinary
UBS Securities Australia Ltd Buy 86,924 8,989 UBS Securities Australia Ltd Buy 11,904 1,567 UBS Securities Australia Ltd Buy 38,303 3,961 UBS Securities Australia Ltd Buy 38,203 3,767 UBS Securities Australia Ltd Buy 5,908 5,178 UBS Securities Australia Ltd Buy 5,908 6,11 UBS Securities Australia Ltd Buy 5,506 6,61 UBS Securities Australia Ltd Buy 5,2433 6,427 UBS Securities Australia Ltd Buy 5,3243 6,427 UBS Securities Australia Ltd Buy 5,3243 6,427 UBS Securities Australia Ltd Buy 5,3243 5,506 UBS Securities Australia Ltd Buy 5,3243 5,506 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 2,29 30 UBS Securities Australia Ltd Buy 339 36 UBS Securities Australia Ltd Buy<	0-Jun-14	UBS Securities Australia Ltd	Buy	11,449	1,184	Ordinary
UBS Securities Australia Ltd Buy 5,483 567 UBS Securities Australia Ltd Buy 11,904 1,231 UBS Securities Australia Ltd Buy 38,303 3,961 UBS Securities Australia Ltd Buy 35,628 5,512 UBS Securities Australia Ltd Buy 5,908 6,112 UBS Securities Australia Ltd Buy 2,578 2,666 UBS Securities Australia Ltd Buy 53,243 6,457 UBS Securities Australia Ltd Buy 53,243 6,457 UBS Securities Australia Ltd Buy 53,243 5,06 UBS Securities Australia Ltd Buy 332 35 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 339 36 UBS Securities Australia Ltd Buy 350 30 UBS Securities Australia Ltd Buy 36 30 UBS Securities Australia Ltd Buy 18 </td <td>0-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>86,924</td> <td>686'8</td> <td>Ordinary</td>	0-Jun-14	UBS Securities Australia Ltd	Buy	86,924	686'8	Ordinary
UBS Securities Australia Ltd Buy 11,904 1,231 UBS Securities Australia Ltd Buy 38,303 3,961 UBS Securities Australia Ltd Buy 36,427 3,767 UBS Securities Australia Ltd Buy 5,908 611 UBS Securities Australia Ltd Buy 5,908 611 UBS Securities Australia Ltd Buy 5,780 2,666 UBS Securities Australia Ltd Buy 53,243 5,506 UBS Securities Australia Ltd Buy 53,243 5,506 UBS Securities Australia Ltd Buy 83,243 5,506 UBS Securities Australia Ltd Buy 83 35 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 290 30 UBS Securities Australia Ltd Buy 290 30 UBS Securities Australia Ltd Buy 290 30 UBS Securities Australia Ltd Buy 184 </td <td>0-lun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>5.483</td> <td>292</td> <td>Ordinary</td>	0-lun-14	UBS Securities Australia Ltd	Buv	5.483	292	Ordinary
UBS Securities Australia Ltd Buy 88,303 3,961 UBS Securities Australia Ltd Buy 36,427 3,767 UBS Securities Australia Ltd Buy 5,908 6,11 UBS Securities Australia Ltd Buy 5,908 6,11 UBS Securities Australia Ltd Buy 62,439 6,657 UBS Securities Australia Ltd Buy 62,439 6,657 UBS Securities Australia Ltd Buy 53,243 5,06 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 220 30 UBS Securities Australia Ltd Buy 230 30 UBS Securities Australia Ltd Buy 136 40 UBS Securities Australia Ltd Buy 136 95 UBS Securities Australia Ltd Buy <td< td=""><td>17</td><td>I IBC Countries Australia I tel</td><td>P. IV</td><td>11 904</td><td>1 231</td><td>Ordinary</td></td<>	17	I IBC Countries Australia I tel	P. IV	11 904	1 231	Ordinary
UBS Securities Australia Ltd Buy 36,437 3,787 UBS Securities Australia Ltd Buy 55,688 5,552 UBS Securities Australia Ltd Buy 5,908 6,11 UBS Securities Australia Ltd Buy 25,008 6,457 UBS Securities Australia Ltd Buy 25,439 6,457 UBS Securities Australia Ltd Buy 5,206 16 UBS Securities Australia Ltd Buy 83,243 5,506 UBS Securities Australia Ltd Buy 83,243 5,506 UBS Securities Australia Ltd Buy 83,243 35 UBS Securities Australia Ltd Buy 339 36 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 136 14 UBS Securities Australia Ltd Buy 136 14 UBS Securities Australia Ltd Buy 136 99	t - 100-0	DOS DECOLUTES AGREEMED END	100	50c 8c	120.0	Ordinary.
UBS Securities Australia Ltd Buy 3,767 UBS Securities Australia Ltd Buy 5,908 5,122 UBS Securities Australia Ltd Buy 5,908 6,11 UBS Securities Australia Ltd Buy 5,2780 2,666 UBS Securities Australia Ltd Buy 53,243 6,427 UBS Securities Australia Ltd Buy 53,243 6,457 UBS Securities Australia Ltd Buy 53,243 6,457 UBS Securities Australia Ltd Buy 53,243 5,506 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 220 30 UBS Securities Australia Ltd Buy 220 30 UBS Securities Australia Ltd Buy 220 30 UBS Securities Australia Ltd Buy 136 40 UBS Securities Australia Ltd Buy 136 50 UBS Securities Australia Ltd Buy 136 9 UBS Securities Australia Ltd Buy 136 9	0-Jun-14	Obs securities Australia Ltd	had	coc,oc	106,0	Clariday
UBS Securities Australia Ltd Buy 5,522 UBS Securities Australia Ltd Buy 5,908 6,11 UBS Securities Australia Ltd Buy 25,780 2,666 UBS Securities Australia Ltd Buy 62,439 6,457 UBS Securities Australia Ltd Buy 53,243 5,506 UBS Securities Australia Ltd Buy 87 9 UBS Securities Australia Ltd Buy 87 9 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 136 14 UBS Securities Australia Ltd Buy 77 8	0-Jun-14	UBS Securities Australia Ltd	Buy	36,427	3,767	Ordinary
URS Securities Australia Ltd Buy 5,908 611 UBS Securities Australia Ltd Buy 25,780 2,666 UBS Securities Australia Ltd Buy 62,439 6,457 UBS Securities Australia Ltd Buy 53,243 5,506 UBS Securities Australia Ltd Buy 87 9 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 73 9 UBS Securities Australia Ltd Buy 77 8	0-Jun-14	UBS Securities Australia Ltd	Buy	53,688	5,552	Ordinary
UBS Securities Australia Ltd Buy 25,780 2,666 UBS Securities Australia Ltd Buy 6,437 6,457 UBS Securities Australia Ltd Buy 52,736 5,066 UBS Securities Australia Ltd Buy 155 16 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 290 30 UBS Securities Australia Ltd Buy 586 99 UBS Securities Australia Ltd Buy 18 14 UBS Securities Australia Ltd Buy 77 8	0-Jun-14	UBS Securities Australia Ltd	Buy	806'5	119	Ordinary
UBS Securities Australia Ltd Buy 62,439 6,457 UBS Securities Australia Ltd Buy 53,243 5,506 UBS Securities Australia Ltd Buy 87 9 UBS Securities Australia Ltd Buy 87 9 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 136 14 UBS Securities Australia Ltd Buy 77 8	0-Jun-14	UBS Securities Australia Ltd	Buv	25,780	2,666	Ordinary
UBS Securities Australia Ltd Buy 53,243 5,506 UBS Securities Australia Ltd Buy 155 16 UBS Securities Australia Ltd Buy 87 33 35 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 290 30 UBS Securities Australia Ltd Buy 296 99 UBS Securities Australia Ltd Buy 184 184 UBS Securities Australia Ltd Buy 77 8	14	I IRC Countries Australia I to	N a	PEN C9	6.457	Ordinary
UBS Securities Australia Ltd Buy 175 100 UBS Securities Australia Ltd Buy 67 9 UBS Securities Australia Ltd Buy 87 9 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 958 99 UBS Securities Australia Ltd Buy 136 14 UBS Securities Australia Ltd Buy 77 8	11 0	De Constitue Australia I to	Bin	567,52	5 506	Ordinary
UBS Securities Australia Ltd Buy 87 99 UBS Securities Australia Ltd Buy 87 99 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 229 30 UBS Securities Australia Ltd Buy 95 99 UBS Securities Australia Ltd Buy 77 8	0-Jul 1-14	UBD Detaillings Augusting Live	, and	10,410	27000	Commany
UBS Securities Australia Ltd Buy 87 9 UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 290 30 UBS Securities Australia Ltd Buy 956 99 UBS Securities Australia Ltd Buy 136 14 UBS Securities Australia Ltd Buy 77 8	0-Jun-14	UBS Securities Australia Lto	Buy	ccl	2 4	Ordinary
UBS Securities Australia Ltd Buy 339 35 UBS Securities Australia Ltd Buy 290 30 UBS Securities Australia Ltd Buy 958 99 UBS Securities Australia Ltd Buy 77 8	0-Jun-14	UBS Securities Australia Ltd	Buy	87	6	Ordinary
UBS Securities Australia Ltd Buy 250 30 UBS Securities Australia Ltd Buy 558 99 UBS Securities Australia Ltd Buy 77 8	0-Jun-14	UBS Securities Australia Ltd	Buy	339	32	Ordinary
UBS Securities Australia Ltd Buy 958 99 UBS Securities Australia Ltd Buy 136 14 UBS Securities Australia Ltd Buy 77 8	2-lun-14	UBS Securities Australia Ltd	Buv	290	30	Ordinary
UBS Securities Australia Ltd Buy 136 14 UBS Securities Australia Ltd Buy 77 8	0-lun-14	UBS Securities Australia Ltd	Buv	856	66	Ordinary
UBS Securities Australia Ltd Buy 77 8	10-11	185 Securities Australia 14d	Riw	136	7.	Ordinary
USS SECURITIES AUSTRAINS LTG BUY 8	1 1 1 1 1	Let Charter Activities Let	,			Ordinan
	0-Jun-14	UBS Securities Australia Liu	Buy	,	20	Ordinary

30-Jun-14					
30-Jun-14	UBS Securities Australia Ltd	Buy	7,066	730	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	232	24	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,865	596	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	5,024	519	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	1,936	200	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	33.861	3.498	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	3.427	354	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buv	4.250	439	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buv	77.440	8.000	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,220	126	Ordinary
Jun-14	UBS Securities Australia Ltd	Buv	9.428	974	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buv	300	31	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	89	7	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buv	1.936	200	Ordinary
Jun-14	UBS Securities Australia Ltd	Biv	1881	6	Ordinary
30-Jun-14	UBS Securities Australia I td	8.5	1 936	2002	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2 904	00.5	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Brv	9 622	700	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	319	33	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	48	2	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	167,425	17,287	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,236	231	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	5,072	524	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Buy	15,159	1,566	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	3,320	343	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,936	200	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	368	38	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	5,518	570	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	319	33	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Buy	1,084	112	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	37,200	3,843	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	136,433	14,087	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	13,898	1,435	Ordinary
-Jun-14	UBS Securities Australia Ltd	Buy	4,695	485	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	232	24	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,507	259	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	6,824	705	Ordinary
10-Jun-14	UBS Securities Australia Ltd	Buy	1,936	200	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	261	27	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	261	27	Ordinary
0-Jun-14	UBS Securities Australia Ltd	Виу	2,091	216	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,104	114	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	242	25	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	896	100	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	4,027	416	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	145,527	15,026	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	16,746	1,730	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,955	202	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	123,406	12,742	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	155	16	Ordinary
30-jun-14	UBS Securities Australia Ltd	Buy	2,304	238	Ordinary
30-fun-14	UBS Securities Australia Ltd	Buy	300	31	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	9,467	978	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	4,308	445	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	2,885	298	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,123	116	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	174	18	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	1,152	119	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	48,400	5,000	Ordinary
10-Jun-14	UBS Securities Australia Ltd	Buy	1,394	144	Ordinary
Jun-14	UBS Securities Australia Ltd	Buy	6,495	671	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	20,183	2,085	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	6,815	704	Ordinary
30-Jun-14	UBS Securities Australia Ltd	Buy	2,981	308	Ordinary
					,
30-Jun-14	UBS Securities Australia Ltd	Buy	25,313	2,615	Ordinary

1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	, ,	``	L	
UBS Securities Australia Ltd	Buy	16,611	1 716	Ordinary
s Australia Ltd	Buy	15,885	1,641	Ordinary
UBS Securities Australia Ltd	Buy	726	75	Ordinary
UBS Securities Australia Ltd	Buy	10,919	1,128	Ordinary
UBS Securities Australia Ltd	Buy	3,872	400	Ordinary
UBS Securities Australia Ltd	Buy	5,343	552	Ordinary
es Australia Ltd	Buy	7,550	780	Ordinary
UBS Securities Australia Ltd	Buy	676	96	Ordinary
UBS Securities Australia Ltd	Buy	318,869	32,941	Ordinary
ies Australia Ltd	ŠŠ	3,030	313	Ordinary
J. IRS Securities Australia 1 td	Biv	186 311	19 247	Ordinary
or Australia 14d	2.0	1646	021	Califord
DO Commission Activities and		1 104	2	O C
thes Australia Ltd	, in a	161,1	2 6	y Serial y
Obs securities Australia Ltd	áng	2,933	sns :	Orginary
UBS Securities Australia Ltd	BUY	3,8/2	400	Ordinary
UBS Securities Australia Ltd	Buy	22,109	2,284	Ordinary
UBS Securities Australia Ltd	Buy	1,258	130	Ordinary
UBS Securities Australia Ltd	Buy	9,361	296	Ordinary
ties Australia Ltd	BUV	17,395	1.797	Ordinary
IRS Securities Australia I to	Biry	2 401	3/18	Ordinary
tres Australia Ltd	ind.	2,70	047	Clumary
Securities Australia Ltd	han	99	4	Ordinary
ities Australia Ltd	Buy	6,418	663	Ordinary
UBS Securities Australia Ltd	Buy	261	27	Ordinary
UBS Securities Australia Ltd	Buy	416	43	Ordinary
rities Australia Ltd	Buv	8.654	894	Ordinary
tioe Americalia I to	Buck	10 736	1 107	Ordina
ries Australia Liu	650	20.00	2,1	Quilling y
Securities Australia Lto	buy	col		Ordinary
JBS Securities Australia Ltd	Buy	861,87	2,913	Ordinary
JBS Securities Australia Ltd	Buy	10,929	1,129	Ordinary
rities Australia Ltd	Buy	113,285	11,703	Ordinary
JBS Securities Australia Ltd	Buy	8,935	673	Ordinary
JBS Securities Australia Ltd	Buy	3,107	321	Ordinary
JBS Securities Australia Ltd	Buv	4.850	501	Ordinary
UBS Securities Australia Ltd	Buv	096.9	719	Ordinary
Societies Australia 14d	BIN	9009	CS	Ordinany
tribe American Lead	6.6	253	3 2	Craining
DOS Seculties Australia Liu	ing	255	000	Cidiliary
DDS Securities Australia Liu	éna e	810,0	0/6	oldinaly.
JBS Securities Australia Ltd	Buy	856	55	Ordinary
JBS Securities Australia Ltd	Buy	3,214	332	Ordinary
UBS Securities Australia Ltd	Buy	4,153	429	Ordinary
rities Australia Ltd	Buy	44,664	4,614	Ordinary
UBS Securities Australia Ltd	Buy	16,214	1,675	Ordinary
rities Australia Ltd	Buv	15.362	1.587	Ordinary
IBS Socirities Australia Ltd	Bis	000 66	2000	Ordinary
white Australia 144	652	059	99	Ordinary
UBS Securities Australia Ltd	eny	650	8	Ordinary
irities Australia Ltd	Buy	33,628	3,474	Ordinary
JBS Securities Australia Ltd	Buy	358	37	Ordinary
JBS Securities Australia Ltd	Buy	4,695	485	Ordinary
JBS Securities Australia Ltd	Buy	1,433	148	Ordinary
ritios Australia I to	Riv	13 320	1 376	Ordinary
IDO Constitute American Lot	Big	10,629	1 000	Ording
ities Australia Eta	ń d	10,02	2000,1	diam's
rities Australia Ltd	Buy	149,6	/66	Ordinary
UBS Securities Australia Ltd	Buy	5,285	546	Ordinary
JBS Securities Australia Ltd	Buy	1,762	182	Ordinary
JBS Securities Australia Ltd	Buy	310	32	Ordinary
Securities Australia 11d	Bir	26	Ç	Ordinary
Con ritios Australia I to	Bus	170	28	Ordinary
Ities Australia Ltd	, and	1/2	07	or or
JBS Securities Australia Ltd	euy -	1,0/4		Ordinary
JBS Securities Australia Ltd	Buy	1,539	159	Ordinary
JBS Securities Australia Ltd	Buy	6,757	869	Ordinary
UBS Securities Australia Ltd	Buy	523	23	Ordinary
UBS Securities Australia Ltd	AN B	871	06	Ordinary
tios Australia I to	20.00	21 151	2 185	Ordinary
ODD DECOLUTES AUSTRALIA ELC	DO.	101,12	20.74	
Aioc Airctralia to		****		(min in i
וונים שתפוו מוומ דונים	Buy	133,392	13,773	Ordinary

	UBS Securities Australia to UB	Buy Buy Buy	436	45	Ordinary Ordinary
	confries Australia Lid Curties Australia Lid Curties Australia Lid Curties Australia Lid Courties Australia Lid Courties Australia Lid Courties Australia Lid Courties Australia Lid Courties Australia Lid Courties Australia Lid	Buy	004	12	Ordinary
	counties Australia Ltd curdines Australia Ltd curdines Australia Ltd counties Australia Ltd counties Australia Ltd counties Australia Ltd counties Australia Ltd counties Australia Ltd counties Australia Ltd counties Australia Ltd counties Australia Ltd counties Australia Ltd				
	courties Australia Ltd cucirities Australia Ltd cucirities Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd cucirities Australia Ltd cucirities Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd	Buv	11.713	1,210	Ordinary
	corrites Australia Ltd curities Australia Ltd curities Australia Ltd curities Australia Ltd curities Australia Ltd curities Australia Ltd curities Australia Ltd curities Australia Ltd curities Australia Ltd	Riv	7878	607	Ordinan
	corrites Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd	Buy	242	75	Ordinary
	confies Australia Ltd confles Australia Ltd confles Australia Ltd confles Australia Ltd confles Australia Ltd confles Australia Ltd	Bix	17 259	1 783	Ordinan
	contries Australia Ltd contries Australia Ltd contries Australia Ltd contries Australia Ltd contries Australia Ltd	Busy	00001	1 062	Ording
	courties Australia Ltd courties Australia Ltd courties Australia Ltd courties Australia Ltd	0	0070	1,002	O CITIES O
	corrities Australia Ltd corrities Australia Ltd	600	902		O Gillian
	contres Australia Ltd	600	90	5,	O G
	curities Australia Ltd	ocy .	0 3	- (Ordinary
1111111	CHILD ALICITATION TO	buy	184	ַ	Ordinary
	Collines messions are	Buy	1,199	124	Ordinary
	UBS Securities Australia Ltd	Buy	3,210	332	Ordinary
	UBS Securities Australia Ltd	Buy	135	14	Ordinary
	UBS Securities Australia Ltd	Buy	445	46	Ordinary
	curities Australia Ltd	Buy	1,083	112	Ordinary
1	UBS Securities Australia Ltd	Buy	71,189	7.358	Ordinary
1 1 1 1 1 1 1 1	Curities Australia Ltd	Buy	096 P6	9815	Ordinary
111111	IRS Securities Australia 14d	Buy	300 67	2,0,0	diam'r.
1	IRC Countries Australia ind	0.00	20077	1001	Ordinary
	Cultings Australia Ltd	ána	616,06	261,6	Ordinary
	UBS Securities Australia Ltd	Buy	45,4/1	4,695	Ordinary
	cunties Australia Ltd	Buy	10,784	1,114	Ordinary
	UBS Securities Australia Ltd	Buy	689'9	169	Ordinary
	UBS Securities Australia Ltd	Buy	5,343	225	Ordinary
Ш	curities Australia Ltd	Buy	16,979	1,754	Ordinary
l	UBS Securities Australia Ltd	Buy	689	691	Ordinary
	Curities Australia I to	18	082 8	acc	Ordingo
1	Curities Australia I to	D. I.	4 165		(Figure 1
30 Lond 14 Library	Carries Australia Lid	909	4,102	430	Orandry
ı	culties Australia Ltd	buy	906	3	Ordinary
1	UBS Securities Australia Ltd	Buy	106	11	Ordinary
٦	UBS Securities Australia Ltd	Buy	6,092	920	Ordinary
30-Jun-14 UBS Se	curities Australia Ltd	Buy	362'E	340	Ordinary
10-Jun-14 UBS Se	curities Australia Ltd	Buv	6.815	704	Ordinary
ı	UBS Securities Australia I to	Buy	1008	55	Ordinary
0-Jun-14 UBS Se	Gurities Australia I td	Biv	11 074	1 144	Ordinary
ı	I IBC Con unities Australia 114	7-0	CUC		1000
	Collines Australia CO	, and	767	07	Ordinary
1	curties Australia Ltd	Buy	153,/80	15,870	Ordinary
1	curities Australia Ltd	Buy	2,674	276	Ordinary
	curities Australia Ltd	Buy	7,074	730	Ordinary
30-Jun-14 UBS Se	UBS Securities Australia Ltd	Buv	63.675	6.578	Ordinary
ł	Outition Australia 14d	Bush	0090	000	
П	Contract Australia Ltd	S S S S S S S S S S S S S S S S S S S	060'6	000,1	Collinally
Ť	DDS Securities Australia Ltd	Buy	55	4	Ordinary
	UBS Securities Australia Ltd	Buy	48	5	Ordinary
	UBS Securities Australia Ltd	Buy	1,638	169	Ordinary
	UBS Securities Australia Ltd	Buv	696	100	Ordinary
10-Jun-14 UBS Sec	curities Australia I td	Ruy	238 452	24 KNR	Ordinary
T	DC Contration Australia	P	120,426	2007.2	(m)
Ť	DO Committee Assessing the	ond ond	0.4.071	200,1	o company
Ť	culties Australia LIU	pan	2,423	750	Orginary
١	UBS Securities Australia Ltd	Buy	11,289	1,165	Ordinary
-1	UBS Securities Australia Ltd	Buy	44,522	4,597	Ordinary
	curities Australia Ltd	Buy	213	22	Ordinary
30-Jun-14 UBS Sec	curities Australia Ltd	Buv	2.258	733	Ordinary
ı	Carities Australia Ltd	2.6	OVE U	622	Ordinan
T	Constitution Programme Little	fina	040,0	700,	0.00
Ť	Culties Australia Ltd	, and	48,276	4,982	Ordinary
┪	JBS Securities Australia Ltd	Buy	68	7	Ordinary
30-Jun-14 UBS Sec	JBS Securities Australia Ltd	Buy	6,269	647	Ordinary
_	curities Australia Ltd	Buv	1.443	149	Ordinary
Ī	UBS Securities Australia Ltd	Buy	5 291	546	Ordinary
T	IBS Segurities Australia I to	(in the second	700 CN	057 7	Ordinany
Ť	IDC Commission Assessed and	bod	766,24	4,439	Coldinary
ı	COLUMN AUSTRALIA	hay	747	57	Ordinary
- 1	curities Australia Ltd	Buy	3,264	337	Ordinary
30-Jun-14 UBS Sec	UBS Securities Australia Ltd	Buy	1,744	180	Ordinary
	curities Australia Ltd	Buv	1.685	174	Ordinary
ł	Curities Australia (1d	Birv	7317	430	Ordinan
30-101-14 00-101-14	Obs Securities Australia Ltd	Buy	4,16/	430	Ordinary

UBS Securities Australia Ltd	bay	5		
	Bitv	242	25	Ordinary
UBS Securities Australia Ltd	Buv	1,481	153	Ordinary
UBS Securities Australia Ltd	Buy	920		Ordinary
UBS Securities Australia Ltd	Buy	1,094	113	Ordinary
3S Securities Australia Ltd	Buy	4,482		Ordinary
UBS Securities Australia Ltd	Buy	10,309		Ordinary
UBS Securities Australia Ltd	Buy	94,196	6	Ordinary
UBS Securities Australia Ltd	Buy	7,366		
UBS Securities Australia Ltd	Buy	2,033		
UBS Securities Australia Ltd	Buy	131,803	13,609	Ordinary
UBS Securities Australia Ltd	Buy	165		
BS Securities Australia Ltd	Buy	1,994	7	
UBS Securities Australia Ltd	Buy	747		Ordinary
UBS Securities Australia Ltd	Buy	832	98	Ordinary
BS Securities Australia Ltd	Buy	7/7		Ordinary
UBS Securities Australia Ltd	Buy	3,891		Ordinary
BS Securities Australia Ltd	Buy	978		
UBS Securities Australia Ltd	Buy	89,056	6	Ordinary
UBS Securities Australia Ltd	Buy	3,872		
UBS Securities Australia Ltd	Buy	6,340		Ordinary
JBS Securities Australia Ltd	Buy	56,280	5,	Ordinary
8S Securities Australia Ltd	Buy	2,323	2	Ordinary
UBS Securities Australia Ltd	Buy	194	20	Ordinary
UBS Securities Australia Ltd	Buy	213		Ordinary
UBS Securities Australia Ltd	Buy	862		Ordinary
UBS Securities Australia Ltd	Buy	29,438		Ordinary
UBS Securities Australia Ltd	Вчу	114,681		
UBS Securities Australia Ltd	Buy	19,380	2,000	
35 Securities Australia Ltd	Buy	1,880		Ordinary
3S Securities Australia Ltd	Buy	37,403		
UBS Securities Australia Ltd	Buy	5,746		
UBS Securities Australia Ltd	Buy	12,597		Ordinary
UBS Securities Australia Ltd	Buy	4,041	417	Ordinary
JBS Securities Australia Ltd	Buy	68		Ordinary
UBS Securities Australia Ltd	Buy	1,260	130	Ordinary
BS Securities Australia Ltd	Buy	606'9		Ordinary
UBS Securities Australia Ltd	Buy	4,118		Ordinary
UBS Securities Australia Ltd	Buy	069'6	1,000	Ordinary
UBS Securities Australia Ltd	Buy	115,211	11,902	Ordinary
JBS Securities Australia Ltd	Buy	186	102	Ordinary
UBS Securities Australia Ltd	Buv	242	25	Ordinary
UBS Securities Australia Ltd	Buv	784	81	Ordinary
IRS Securities Australia Ltd	Bir	1 374		Vienipi
IIIC Coourities Australia Ltd	A Control	89		Company
Do Securifies Adstralia Lid	, and	2316		O-diminally
ODS Securities Australia Ltd	ina	2,210	677	Ordinary
UBS Securities Australia Ltd	eny	566,11		Orginary
BS securities Australia Ltd	Buy	/68'1		Ordinary
UBS Securities Australia Ltd	Buy	1,384	143	Ordinary
IBS Securities Australia Ltd	Buy	4,/43	490	Ordinary
UBS Securities Australia Ltd	Buy	106		Ordinary
UBS Securities Australia Ltd	Buy	6,8/3	01/	Ordinary
85 Securities Australia Ltd	ana	430	45	Ordinary
JBS Securities Australia Ltd	Buy	6/9/9	080	Ordinary
UBS Securities Australia Ltd	eny	338	75.	Ordinary
BS securities Australia Lto	any.	306	001	Ordinary
UBS Securities Australia Ltd	Śng -	74/47	900'7	Ordinary
OS Coomition Authority 14d	600	346	26	Ordinaly
Upo Committee Australia Ltd	sing l	3 673	30	Ordinary
UBS Securities Australia Ltd	Buy	3,8/2	004	Ordinary
UBS Securities Australia Ltd	Buy	5/9,1	1/3	Ordinary
UBS Securities Australia Ltd	Buy	213	22	Ordinary
BS Securities Australia Ltd	Buy	2,904	900	Ordinary
UBS Securities Australia Ltd	Buy	155	16	Ordinary
UBS Securities Australia Ltd	Buv	407	77	Ordinary
	600			Oldina)

105 Securine Australia Ltd Buy \$2,47 \$40,78 105 Securine Australia Ltd Buy \$6,51 \$40,78 \$6,61 \$10,78 105 Securine Australia Ltd Buy \$6,54 \$6,78 \$7,70 \$6,62 \$10,85 105 Securine Australia Ltd Buy \$6,54 \$6,78 \$10,70	0-Jun 14	CONTRACTOR SECTION OF THE CONTRACTOR SECTION				
OS Securities Australia LID Buy 6,50 OS Securities Australia LID Buy 6,60 OS Securities Australia LID Buy 6,60 OS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 6,50 LOS Securities Australia LID Buy 7,73	200	I IDC Consistion Ametralia 14d	Buy Bus	242	25	Ordinary
USS Securities Australia Ltd Buy C6,578 C6,245 USS Securities Australia Ltd Buy C6,578 C6,245 USS Securities Australia Ltd Buy C6,578 C6,245 USS Securities Australia Ltd Buy C6,578 C6,245 USS Securities Australia Ltd Buy C6,578 C6,245 USS Securities Australia Ltd Buy C7,774 C7,774 C7,774 USS Securities Australia Ltd Buy C7,774	0-Jun-14	UBS Securities Australia I to	Sus.	036,916	4,078	Ordinary
OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy 65/20 OSS Securities Australia LIV Buy	20-fun-14	I IRS Societies Australia 144	D. C.	000	05 55	Ordinary
UBS Securities Australia Ltd Buy 6,547 6,548 UBS Securities Australia Ltd Buy 6,547 2,64 UBS Securities Australia Ltd Buy 6,547 2,64 UBS Securities Australia Ltd Buy 2,713 2,81 UBS Securities Australia Ltd Buy 2,074 2,82 UBS Securities Australia Ltd Buy 2,52 2,63 UBS Securities Australia Ltd Buy 4,75 4,75 UBS Securities Australia Ltd Buy 4,75 4,75 UBS Securities Australia Ltd Buy 4,75 4,75 UBS Securities Australia Ltd Buy 2,75 3,78 UBS Securities Australia Ltd Buy 2,75 4,75 UBS Securities Australia Ltd Buy	10000	ODS Securities Australia Ltd	, and	46,439	4,/90	Ordinary
UBS Securities Australia Ltd Buy 6,5,5,7 6,2,3 UBS Securities Australia Ltd Buy 5,2,9 1,0,3 UBS Securities Australia Ltd Buy 2,5,4 2,5,1 UBS Securities Australia Ltd Buy 2,5,2 2,5 UBS Securities Australia Ltd Buy 4,5,5 4,7 UBS Securities Australia Ltd Buy 4,5,5 4,5 UBS Securities Australia Ltd Buy	0-30U-14	Ubb Securities Australia Ltd	Buy	8/9'69	7,187	Ordinary
March March May 10,558 566 168 168 168 168 168 169	0-Jun-14	UBS Securities Australia Ltd	Buy	60,547	6,242	Ordinary
UBS Securities Australia Ltd Blay 10,558 1,09 UBS Securities Australia Ltd Blay 1,05 1,09 UBS Securities Australia Ltd Blay 2,514 2,61 UBS Securities Australia Ltd Blay 2,514 2,63 UBS Securities Australia Ltd Blay 2,514 2,63 UBS Securities Australia Ltd Blay 2,614 2,614 UBS Securities Australia Ltd Blay 4,154 4,13 UBS Securities Australia Ltd Blay 4,154 4,13 UBS Securities Australia Ltd Blay 5,63 5,50 UBS Securities Australia Ltd Blay 5,63 5,63 UBS Securities Australia Ltd Blay 5,63 2,78 UBS Securities Australia Ltd Blay 5,63 2,78 UBS Securities Australia Ltd Blay 5,63 1,49 UBS Securities Australia Ltd Blay 7,154 1,49 UBS Securities Australia Ltd Blay 7,154 1,149 UBS Securities Australia Ltd	30-Jun-14	UBS Securities Australia Ltd	Buy	5,291	546	Ordinary
UBS Securities Australia Ltd Buy 68.0 88.0 UBS Securities Australia Ltd Buy 27.174 28.1 UBS Securities Australia Ltd Buy 27.174 2.81.3 UBS Securities Australia Ltd Buy 2.7.40 2.81.47 UBS Securities Australia Ltd Buy 2.0.740 2.1.47 UBS Securities Australia Ltd Buy 2.0.740 2.1.47 UBS Securities Australia Ltd Buy 2.0.7 4.8.804 UBS Securities Australia Ltd Buy 4.1.82 6.0.7 UBS Securities Australia Ltd Buy 4.1.83 6.0.7 UBS Securities Australia Ltd Buy 4.1.83 6.0.7 UBS Securities Australia Ltd Buy 4.1.83 1.1.89 UBS Securities Australia Ltd Buy 4.1.83 1.1.89 UBS Securities Australia Ltd Buy 5.2.99 4.0.9 UBS Securities Australia Ltd Buy 7.1.83 1.1.89 UBS Securities Australia Ltd Buy 7.1.82 1.0.9 UBS Securities Austra	30-Jun-14	UBS Securities Australia Ltd	Buy	10,558	1,093	Ordinary
UBS Securities Australia Ltd Buy 261 27 UBS Securities Australia Ltd Buy 20,747 281 UBS Securities Australia Ltd Buy 20,740 2,843 UBS Securities Australia Ltd Buy 20,740 4,804 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,50 4,73 UBS Securities Australia Ltd Buy 2,00 4,73 UBS Securities Australia Ltd Buy	0-Jun-14	UBS Securities Australia Ltd	Buy	058	88	Ordinary
UBS Securities Australia Ltd Buy 27,174 2815 UBS Securities Australia Ltd Buy 27,141 281 UBS Securities Australia Ltd Buy 20,440 2,147 UBS Securities Australia Ltd Buy 20,740 2,147 UBS Securities Australia Ltd Buy 5,893 610 UBS Securities Australia Ltd Buy 4,143 4,143 UBS Securities Australia Ltd Buy 4,143 4,143 UBS Securities Australia Ltd Buy 4,143 4,143 UBS Securities Australia Ltd Buy 4,153 4,143 UBS Securities Australia Ltd Buy 4,159 4,159 UBS Securities Australia Ltd Buy 4,159 4,154 UBS Securities Australia Ltd Buy 4,159 4,159 UBS Securities Australia Ltd Buy 5,299 3,09 UBS Securities Australia Ltd Buy 7,178 1,149 UBS Securities Australia Ltd Buy 7,128 1,149 UBS Securities Australia Ltd <th< td=""><td>0-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>261</td><td>77</td><td>Ordinary</td></th<>	0-Jun-14	UBS Securities Australia Ltd	Buy	261	77	Ordinary
UBS Securities Australia Ltd Bay 2,247 2,28 UBS Securities Australia Ltd Bluy 2,247 2,28 UBS Securities Australia Ltd Bluy 2,247 4,147 UBS Securities Australia Ltd Bluy 4,154 4,154 UBS Securities Australia Ltd Bluy 9,667 1,00 UBS Securities Australia Ltd Bluy 9,678 4,154 UBS Securities Australia Ltd Bluy 9,678 4,154 UBS Securities Australia Ltd Bluy 4,573 4,73 UBS Securities Australia Ltd Bluy 4,579 4,73 UBS Securities Australia Ltd Bluy 4,599 4,73 UBS Securities Australia Ltd Bluy 4,599 4,73 UBS Securities Australia Ltd Bluy 2,008 1,700 UBS Securities Australia Ltd Bluy 1,108 1,109 UBS Securities Australia Ltd Bluy 2,108 1,109 UBS Securities Australia Ltd Bluy 2,108 1,109 UBS Securities Australia Ltd	0-lun-14	UBS Securities Australia I td	Bin	171 75	2 643	Contract
UBS Securities Australia Ltd Buy 20,740 2,147 UBS Securities Australia Ltd Buy 20,740 4,860 UBS Securities Australia Ltd Buy 20,741 4,860 UBS Securities Australia Ltd Buy 5,893 610 UBS Securities Australia Ltd Buy 4,890 4,300 UBS Securities Australia Ltd Buy 4,589 473 UBS Securities Australia Ltd Buy 4,589 4,78 UBS Securities Australia Ltd Buy 7,788 1,78 UBS Securities Australia Ltd Buy 7,789 1,78 UBS Securities Australia Ltd Buy 2,299 1,48 UBS Securities Australia Ltd Buy	0-lim-14	I IBS Sequifies Australia I to		27.7.7	2,012	Cidinary
UBS Securities Australia Ltd Buy 420,740 UBS Securities Australia Ltd Buy 45,24 UBS Securities Australia Ltd Buy 5,335 UBS Securities Australia Ltd Buy 6,675 UBS Securities Australia Ltd Buy 6,678 UBS Securities Australia Ltd Buy 7,103 UBS Securities Australia Ltd Buy	14	I IDC Cognition Australia I th	600	1,45,2	507	Ordinary
UBS Securities Australia Ltd BMy 442,807 44,804 UBS Securities Australia Ltd BMy 582,807 46,000 UBS Securities Australia Ltd BMy 5,893 610 UBS Securities Australia Ltd BMy 6,675 691 UBS Securities Australia Ltd BMy 6,675 691 UBS Securities Australia Ltd BMy 4,589 473 UBS Securities Australia Ltd BMy 4,589 473 UBS Securities Australia Ltd BMy 4,789 473 UBS Securities Australia Ltd BMy 4,789 4,780 UBS Securities Australia Ltd BMy 2,929 3,033 UBS Securities Australia Ltd BMy 1,349 1,502 UBS Securities Australia Ltd BMy 1,349 1,524 1,002 UBS Securities Australia L	111111111111111111111111111111111111111	Obs securities Abstralia Lid	had	20,740	7,14/	Ordinary
UBS Securities Australia Ltd Bluy 965 100 UBS Securities Australia Ltd Bluy 566 100 UBS Securities Australia Ltd Bluy 5,335 652 UBS Securities Australia Ltd Bluy 5,335 652 UBS Securities Australia Ltd Bluy 6,673 691 UBS Securities Australia Ltd Bluy 6,673 692 UBS Securities Australia Ltd Bluy 6,673 691 UBS Securities Australia Ltd Bluy 6,673 6,773 UBS Securities Australia Ltd Bluy 6,520 6,773 UBS Securities Australia Ltd Bluy 7,670 1,149 UBS Securities Australia Ltd Bluy 7,580 1,69 UBS Securities Australia Ltd Bluy<	0-Jun-14	UBS Securities Australia Ltd	Buy	432,807	44,804	Ordinary
UBS Securities Australia Ltd Buy 4,154 UBS Securities Australia Ltd Buy 4,154 UBS Securities Australia Ltd Buy 1,230 1,231 UBS Securities Australia Ltd Buy 5,883 610 UBS Securities Australia Ltd Buy 6,675 691 UBS Securities Australia Ltd Buy 4,289 473 UBS Securities Australia Ltd Buy 7,788 1,770 UBS Securities Australia Ltd Buy 7,788 1,89 UBS Securities Australia Ltd Buy 7,786 1,89 UBS Securities Australia Ltd Buy 7,786 1,80 UBS Securities Australia Ltd Buy 7,786 1,80 <td>0-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>362</td> <td>100</td> <td>Ordinary</td>	0-Jun-14	UBS Securities Australia Ltd	Buy	362	100	Ordinary
UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 5,832 6,53 UBS Securities Australia Ltd Buy 6,673 6,91 UBS Securities Australia Ltd Buy 6,673 4,93 6,91 UBS Securities Australia Ltd Buy 6,673 4,63 6,91 UBS Securities Australia Ltd Buy 6,629 4,73 1,79 UBS Securities Australia Ltd Buy 6,629 4,73 4,73 6,63 UBS Securities Australia Ltd Buy 6,629 4,73 4,73 6,63 UBS Securities Australia Ltd Buy 7,107 1,70 1,70 1,70 UBS Securities Australia Ltd Buy 7,10 1,70 1,70 1,70 UBS Securities Australia Ltd Buy 7,10 1,70 1,70 1,70 UBS Securities Australia Ltd Buy 7,10 1,70 1,70 1,70 UBS Securities Australia Ltd Buy 1,70 1,70 1,70	0-Jun-14	UBS Securities Australia Ltd	Buy	996	100	Ordinary
UBS Securities Australia Ltd Buy 5,893 610 UBS Securities Australia Ltd Buy 5,330 1,791 UBS Securities Australia Ltd Buy 6,675 691 UBS Securities Australia Ltd Buy 4,589 473 UBS Securities Australia Ltd Buy 4,589 475 UBS Securities Australia Ltd Buy 4,589 1,700 UBS Securities Australia Ltd Buy 4,132 1,40 UBS Securities Australia Ltd Buy 4,132 1,14 UBS Securities Australia Ltd Buy 1,708 1,14 UBS Securities Australia Ltd Buy	0-Jun-14	UBS Securities Australia Ltd	Buy	4,154	430	Ordinary
UBS Securities Australia Ltd Bby 17371 1737 UBS Securities Australia Ltd Buy 6,573 5,53 UBS Securities Australia Ltd Buy 6,573 6,573 UBS Securities Australia Ltd Buy 4,784 4,73 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 2,1078 2,878 UBS Securities Australia Ltd Buy 2,1078 1,700 UBS Securities Australia Ltd Buy 1,700 1,700 UBS Securities Australia Ltd	:0-Jun-14	UBS Securities Australia Ltd	Buy	5.893	610	Ordinary
UBS Securities Australia Ltd Bay 5,332 5,53 UBS Securities Australia Ltd Buy 6,675 6,91 UBS Securities Australia Ltd Buy 4,599 475 UBS Securities Australia Ltd Buy 4,599 475 UBS Securities Australia Ltd Buy 4,599 473 UBS Securities Australia Ltd Buy 4,599 4,79 UBS Securities Australia Ltd Buy 4,599 4,79 UBS Securities Australia Ltd Buy 2,090 6,78 UBS Securities Australia Ltd Buy 2,100 2,182 UBS Securities Australia Ltd Buy 2,100 2,182 UBS Securities Australia Ltd Buy 1,150 1,149 UBS Securities Australia Ltd Buy 1,158 1,158 UBS Securities Australia Ltd Buy	0-jun-14	UBS Securities Australia Ltd	Briv	17 301	1 701	Ordingo
UBS Securities Australia Ltd Buy 6,575 532 UBS Securities Australia Ltd Buy 4,743 491 UBS Securities Australia Ltd Buy 4,743 491 UBS Securities Australia Ltd Buy 4,579 4,73 UBS Securities Australia Ltd Buy 2,624 2,62 UBS Securities Australia Ltd Buy 2,1078 2,182 UBS Securities Australia Ltd Buy 2,192 1,09 UBS Securities Australia Ltd Buy 2,192 1,09 UBS Securities Australia Ltd Buy 2,193 1,09 UBS Securities Australia Ltd Buy 2,194 1,00 UBS Securities Australia Ltd Buy 2,194 1,00 UBS Securities Australia Ltd Buy </td <td>0-4 m-14</td> <td>I IBS Securities Australia 14d</td> <td>, in a</td> <td>100,00</td> <td>16,7,1</td> <td>O COLUMN</td>	0-4 m-14	I IBS Securities Australia 14d	, in a	100,00	16,7,1	O COLUMN
UBS Securities Australia Ltd Buy 4,75 691 UBS Securities Australia Ltd Buy 4,733 4,91 UBS Securities Australia Ltd Buy 4,589 4,59 UBS Securities Australia Ltd Buy 4,599 4,599 UBS Securities Australia Ltd Buy 4,599 1,700 UBS Securities Australia Ltd Buy 4,599 1,700 UBS Securities Australia Ltd Buy 2,700 2,878 UBS Securities Australia Ltd Buy 2,700 2,878 UBS Securities Australia Ltd Buy 1,709 1,149 UBS Securities Australia Ltd Buy 1,708 1,700 UBS Securities Australia Ltd Buy 1,708 1,800 UBS Securities Australia Ltd Bu		The Consider Assessed Like	, and	2,532	766	Ordinary
UBS Securities Australia Ltd Buy 4,743 491 UBS Securities Australia Ltd Buy 4,579 475 UBS Securities Australia Ltd Buy 4,579 475 UBS Securities Australia Ltd Buy 4,579 4,78 UBS Securities Australia Ltd Buy 2,634 2,678 UBS Securities Australia Ltd Buy 2,708 1,700 UBS Securities Australia Ltd Buy 2,708 1,700 UBS Securities Australia Ltd Buy 2,708 1,700 UBS Securities Australia Ltd Buy 2,786 1,700 UBS Securities Australia Ltd Buy 1,756 1,032 UBS Securities Australia Ltd Buy 7,786 1,002 UBS Securities Australia Ltd Buy<	111111111111111111111111111111111111111	ODD DECORINES AUSTRAIR LIQ	Buy	6/9/9	169	Ordinary
URS Securities Australia Ltd Buy 4,589 475 URS Securities Australia Ltd Buy 4,579 474 URS Securities Australia Ltd Buy 6,549 6,549 URS Securities Australia Ltd Buy 26,734 6,78 URS Securities Australia Ltd Buy 27,7801 2,878 URS Securities Australia Ltd Buy 27,7801 2,878 URS Securities Australia Ltd Buy 2,929 3,033 URS Securities Australia Ltd Buy 2,929 3,033 URS Securities Australia Ltd Buy 1,788 1,149 URS Securities Australia Ltd Buy 1,788 1,82 URS Securities Australia Ltd Buy 1,788 1,82 URS Securities Australia Ltd Buy 1,786 1,149 URS Securities Australia Ltd <th< td=""><td>0-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>4,743</td><td>491</td><td>Ordinary</td></th<>	0-Jun-14	UBS Securities Australia Ltd	Buy	4,743	491	Ordinary
URS Securities Australia Ltd Buy 457.9 47.4 URS Securities Australia Ltd Buy 26.034 2.655 URS Securities Australia Ltd Buy 26.034 2.657 URS Securities Australia Ltd Buy 2.670 2.878 URS Securities Australia Ltd Buy 2.70 2.83 URS Securities Australia Ltd Buy 2.70 2.92.29 3.03 URS Securities Australia Ltd Buy 2.92.29 3.03 URS Securities Australia Ltd Buy 2.92.29 3.03 URS Securities Australia Ltd Buy 7.786 8.86 URS Securities Australia Ltd Buy 7.786 8.87 URS Securities Australia Ltd Buy 7.736 1.87 1.00 URS Securities Australia Ltd Buy 7.736 1.87 1.1	0-Jun-14	UBS Securities Australia Ltd	Buy	4,589	475	Ordinary
UBS Securities Australia Ltd Buy 18,335 1,896 UBS Securities Australia Ltd Buy 26,034 2,695 UBS Securities Australia Ltd Buy 26,034 2,695 UBS Securities Australia Ltd Buy 27,001 2,878 UBS Securities Australia Ltd Buy 27,008 1,770 UBS Securities Australia Ltd Buy 2,929 3,033 UBS Securities Australia Ltd Buy 2,929 1,700 UBS Securities Australia Ltd Buy 1,729 1,149 UBS Securities Australia Ltd Buy 1,529 1,149 UBS Securities Australia Ltd Buy 1,528 1,639 UBS Securities Australia Ltd Buy 1,736 1,748 1,748 UBS Securities Australia Ltd Buy 1,736 1,749 1,749 UBS Securities Australia Ltd Buy 1,736 1,749 1,749 UBS Securities Australia Ltd Buy 1,736 1,749 1,749 UBS Securities Australia Ltd Buy 1,736	0-Jun-14	UBS Securities Australia Ltd	Buv	4 579	474	Ordinary
UBS Securities Australia Ltd Buy 25,034 2,658 UBS Securities Australia Ltd Buy 25,034 2,658 UBS Securities Australia Ltd Buy 2,1078 2,182 UBS Securities Australia Ltd Buy 2,7,601 2,182 UBS Securities Australia Ltd Buy 2,7,601 2,878 UBS Securities Australia Ltd Buy 2,7,601 2,878 UBS Securities Australia Ltd Buy 2,929 3,38 UBS Securities Australia Ltd Buy 2,929 3,38 UBS Securities Australia Ltd Buy 1,788 1,82 UBS Securities Australia Ltd Buy 1,788 1,82 UBS Securities Australia Ltd Buy 1,788 1,82 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,786 1,00 UBS Securities Australia Ltd Buy 7,113 1,00 UBS Securities Australia Ltd Buy 7,113 1,00 UBS Securities Australia Ltd <t< td=""><td>0-Jun-14</td><td>UBS Securities Australia Itd</td><td>B</td><td>355 81</td><td>1 909</td><td>College</td></t<>	0-Jun-14	UBS Securities Australia Itd	B	355 81	1 909	College
UBS Securities Australia Ltd Buy COLUSA 4.05 UBS Securities Australia Ltd Buy 6,549 6.78 4.05 UBS Securities Australia Ltd Buy 21,026 1,770 2,878 4.00 UBS Securities Australia Ltd Buy 21,029 3,033 4.00 4.00 2,107 2,878 4.00	1 - 0 - 1 - 0	I IBC Countries Australia 14d	600	555 SC	060'1	Crainary
UBS Securities Australia Ltd Buy 6.549 6.78 UBS Securities Australia Ltd Buy 6.549 6.78 UBS Securities Australia Ltd Buy 7.1078 2.182 UBS Securities Australia Ltd Buy 4.154 4.30 UBS Securities Australia Ltd Buy 4.154 4.30 UBS Securities Australia Ltd Buy 4.154 4.30 UBS Securities Australia Ltd Buy 2.22,29 3.03 UBS Securities Australia Ltd Buy 1.149 1.049 UBS Securities Australia Ltd Buy 1.788 1.82 UBS Securities Australia Ltd Buy 7.189 1.82 UBS Securities Australia Ltd Buy 7.788 1.86 UBS Securities Australia Ltd Buy 7.113 7.26 UBS Securities Australia Ltd Buy <td>11 1</td> <td>The Committee According and</td> <td>acy</td> <td>20,034</td> <td>C69'7</td> <td>Orainary</td>	11 1	The Committee According and	acy	20,034	C69'7	Orainary
UBS Securities Australia Ltd Buy 57.80 UBS Securities Australia Ltd Buy 27.801 2.182 UBS Securities Australia Ltd Buy 27.008 2.182 UBS Securities Australia Ltd Buy 1.709 1.770 UBS Securities Australia Ltd Buy 1.752 1.40 UBS Securities Australia Ltd Buy 1.758 1.709 UBS Securities Australia Ltd Buy 1.758 1.78 UBS Securities Australia Ltd Buy 1.758 1.092 UBS Securities Australia Ltd Buy 1.788 1.893 1.96 UBS Securities Australia Ltd Buy 1.788 1.89 1.78 1.09 UBS Securities Australia Ltd Buy 1.786 1.893 1.96 1.1 1.1 1.1 1.1 1.1 1.0 1.1 1.0 1.0 1.1 1.0 1.1 1.0 1.1 1.0 1.0 1.1 1.0 1.0 1.1 1.0 1.0 1.0 1.0 1.0 1	0-Jun-14	UBS Securities Australia Ltd	Buy	406	42	Ordinary
UBS Securities Australia Ltd Buy 27,801 2,878 UBS Securities Australia Ltd Buy 21,078 1,709 UBS Securities Australia Ltd Buy 1,522 1,40 UBS Securities Australia Ltd Buy 23,239 3,033 UBS Securities Australia Ltd Buy 22,239 3,033 UBS Securities Australia Ltd Buy 23,249 3,033 UBS Securities Australia Ltd Buy 23,249 1,049 UBS Securities Australia Ltd Buy 1,788 182 UBS Securities Australia Ltd Buy 7,786 186 UBS Securities Australia Ltd Buy 7,113 7,36 UBS Securities Australia Ltd Buy 1,314 1,36 UBS Securities Australia Ltd Buy 2,13 2,2 UBS Securities Australia Ltd Buy<	0-Jun-14	UBS Securities Australia Ltd	Buy	6,549	829	Ordinary
UBS Securities Australia Ind Buy 21,078 2,182 UBS Securities Australia Ltd Buy 1,770 140 UBS Securities Australia Ltd Buy 1,322 140 UBS Securities Australia Ltd Buy 1,322 1,43 UBS Securities Australia Ltd Buy 1,059 1,149 UBS Securities Australia Ltd Buy 1,786 186 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 2,13 2,14 UBS Securities Australia Ltd Buy 2,13 2,14 UBS Securities Australia Ltd Buy	0-Jun-14	UBS Securities Australia Ltd	Buy	108'22	2,878	Ordinary
UBS Securities Australia Ltd Buy 17,098 1,770 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 1,329 3,033 UBS Securities Australia Ltd Buy 1,099 1,149 UBS Securities Australia Ltd Buy 1,059 1,149 UBS Securities Australia Ltd Buy 1,059 1,149 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,13 736 UBS Securities Australia Ltd Buy 3,20 3,396 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,213 4,14 UBS Securities Australia Ltd Buy <	0-Jun-14	UBS Securities Australia Ltd	Buv	21.078	2 182	Ordinary
UBS Securities Australia Ltd Buy 4,154 4,10 UBS Securities Australia Ltd Buy 4,154 4,10 UBS Securities Australia Ltd Buy 29,299 3,033 UBS Securities Australia Ltd Buy 6,598 6,83 UBS Securities Australia Ltd Buy 10,549 1,092 UBS Securities Australia Ltd Buy 1,788 182 UBS Securities Australia Ltd Buy 7,786 10 UBS Securities Australia Ltd Buy 7,18 12 UBS Securities Australia Ltd Buy 7,18 3,36 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,33 4,14 UBS Securities Australia Ltd Buy 4,15	0-lun-14	I IRS Securities Australia 11d	Rev	17 000	1770	Carle C
UBS Securities Australia Ltd Buy 1,124 430 UBS Securities Australia Ltd Buy 1,929 3,033 UBS Securities Australia Ltd Buy 1,092 3,033 UBS Securities Australia Ltd Buy 1,0549 1,092 UBS Securities Australia Ltd Buy 1,786 180 UBS Securities Australia Ltd Buy 7,786 180 UBS Securities Australia Ltd Buy 7,786 101 UBS Securities Australia Ltd Buy 7,786 101 UBS Securities Australia Ltd Buy 7,786 101 UBS Securities Australia Ltd Buy 7,713 756 UBS Securities Australia Ltd Buy 1,000 1,000 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy <	0-1 10-14	1 BC Counties Australia 14d	D	2.17	2//	- I
UBS Securities Australia Ltd Buy 29,299 3 (13) UBS Securities Australia Ltd Buy 6,598 6,633 UBS Securities Australia Ltd Buy 1,149 1,149 UBS Securities Australia Ltd Buy 1,758 182 UBS Securities Australia Ltd Buy 7,788 182 UBS Securities Australia Ltd Buy 7,788 186 UBS Securities Australia Ltd Buy 7,113 156 UBS Securities Australia Ltd Buy 7,113 156 UBS Securities Australia Ltd Buy 7,113 176 11 UBS Securities Australia Ltd Buy 38,022 3,336 14 UBS Securities Australia Ltd Buy 1,314 136 136 UBS Securities Australia Ltd Buy 22 3,336 144 14 UBS Securities Australia Ltd Buy 4,154 4,30 1,000 1,000 UBS Securities Australia Ltd Buy 4,154 4,30 1,000 1,000 1,000	14	Lat Countries A society of OCI	65	tr-'t	000	Courary
UBS Securities Australia Ltd Buy 6289 633 UBS Securities Australia Ltd Buy 1,149 1,149 UBS Securities Australia Ltd Buy 1,149 1,149 UBS Securities Australia Ltd Buy 1,758 188 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,713 736 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 1,014 10 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 430 UBS Securities Australia Ltd Buy 1,324 430 UBS Securities Australia Ltd Buy 4,15	11000	COO Decar lines Australia Eto	BOX	755,	140	Ordinary
UBS Securities Australia Ltd Buy 6,598 683 UBS Securities Australia Ltd Buy 11,099 1,149 UBS Securities Australia Ltd Buy 1,1758 182 UBS Securities Australia Ltd Buy 7,786 180 UBS Securities Australia Ltd Buy 7,786 196 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 38,022 3,336 UBS Securities Australia Ltd Buy 38,022 3,336 UBS Securities Australia Ltd Buy 2,23 3,44 4,336 UBS Securities Australia Ltd Buy 2,53 4,14 4,134 4,136 UBS Securities Australia Ltd Buy 8,56 1,000 1,000 1,000 UBS Securities Australia Ltd Buy 4,154 4,30 1,000 1,000 UBS Securities Australia Ltd </td <td>0-JUN-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>29,299</td> <td>3,033</td> <td>Ordinary</td>	0-JUN-14	UBS Securities Australia Ltd	Buy	29,299	3,033	Ordinary
UBS Securities Australia Ltd Buy 11,099 1,149 UBS Securities Australia Ltd Buy 10,549 1,092 UBS Securities Australia Ltd Buy 7,786 182 UBS Securities Australia Ltd Buy 7,786 106 UBS Securities Australia Ltd Buy 7,786 101 UBS Securities Australia Ltd Buy 7,713 736 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,33 3,956 UBS Securities Australia Ltd Buy 2,34 4,15 UBS Securities Australia Ltd Buy 2,34 4,16 UBS Securities Australia Ltd Buy 2,13 2,2 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy <td< td=""><td>0-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>6,598</td><td>683</td><td>Ordinary</td></td<>	0-Jun-14	UBS Securities Australia Ltd	Buy	6,598	683	Ordinary
UBS Securities Australia Ltd Buy 10,549 1,092 UBS Securities Australia Ltd Buy 3,149 326 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,786 106 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,23 396 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,1	0-Jun-14	UBS Securities Australia Ltd	Buy	11,099	1,149	Ordinary
UBS Securities Australia Ltd Buy 3,449 3,66 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,713 15 UBS Securities Australia Ltd Buy 7,113 7,13 UBS Securities Australia Ltd Buy 7,113 7,13 UBS Securities Australia Ltd Buy 7,113 1,36 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,29 3,49 UBS Securities Australia Ltd Buy 2,29 3,49 UBS Securities Australia Ltd Buy 2,29 3,49 UBS Securities Australia Ltd Buy 2,13 3,20 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy	0-Jun-14	UBS Securities Australia Ltd	Buy	10 549	1 092	Ordinary
UBS Securities Australia Ltd Buy 1,78 182 UBS Securities Australia Ltd Buy 1,786 806 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 976 101 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,33 3 UBS Securities Australia Ltd Buy 2,314 136 UBS Securities Australia Ltd Buy 2,34 4,30 UBS Securities Australia Ltd Buy 2,34 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 <td>0-3-14</td> <td>I IBS Securities Australia 14d</td> <td>0</td> <td>0770</td> <td>700/</td> <td>, ,</td>	0-3-14	I IBS Securities Australia 14d	0	0770	700/	, ,
UBS Securities Australia Ltd Buy 7,786 182 UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 7,113 15 UBS Securities Australia Ltd Buy 976 111 UBS Securities Australia Ltd Buy 976 111 UBS Securities Australia Ltd Buy 3,002 3,336 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 22 1,314 136 UBS Securities Australia Ltd Buy 2,13 3,20 1,000 UBS Securities Australia Ltd Buy 2,13 3,4 1,000 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Se	1 1 1	COD OCCUPATION AUGUST AND	Buy	0,140	075	Ordinary
UBS Securities Australia Ltd Buy 7,786 806 UBS Securities Australia Ltd Buy 1,738 196 UBS Securities Australia Ltd Buy 145 115 UBS Securities Australia Ltd Buy 7,113 738 UBS Securities Australia Ltd Buy 7,113 736 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,33 34 UBS Securities Australia Ltd Buy 2,314 136 UBS Securities Australia Ltd Buy 2,33 34 UBS Securities Australia Ltd Buy 2,56 1,000 UBS Securities Australia Ltd Buy 3,56 4,14 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 <td>- Income</td> <td>Obs secontiles Australia Ltd</td> <td>Buy</td> <td>1,758</td> <td>182</td> <td>Ordinary</td>	- Income	Obs secontiles Australia Ltd	Buy	1,758	182	Ordinary
UBS Securities Australia Ltd Buy 1833 196 UBS Securities Australia Ltd Buy 145 115 UBS Securities Australia Ltd Buy 976 111 UBS Securities Australia Ltd Buy 7,13 736 UBS Securities Australia Ltd Buy 38,022 3,396 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,23 3,996 UBS Securities Australia Ltd Buy 2,13 3,2 UBS Securities Australia Ltd Buy 2,13 3,2 UBS Securities Australia Ltd Buy 2,13 3,4 UBS Securities Australia Ltd Buy 2,13 3,2 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 <td>0-Jun-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>7,786</td> <td>908</td> <td>Ordinary</td>	0-Jun-14	UBS Securities Australia Ltd	Buy	7,786	908	Ordinary
UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 976 101 UBS Securities Australia Ltd Buy 7,13 75 UBS Securities Australia Ltd Buy 10 1 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,13 23 UBS Securities Australia Ltd Buy 2,13 34 UBS Securities Australia Ltd Buy 8,568 887 UBS Securities Australia Ltd Buy 1,565 1,000 UBS Securities Australia Ltd Buy 1,565 1,000 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154	0-Jun-14	UBS Securities Australia Ltd	Buy	1,893	196	Ordinary
UBS Securities Australia Ltd Buy 7.113 7.36 UBS Securities Australia Ltd Buy 7.113 7.36 UBS Securities Australia Ltd Buy 106 1.1 UBS Securities Australia Ltd Buy 1.314 1.36 UBS Securities Australia Ltd Buy 1.314 1.36 UBS Securities Australia Ltd Buy 2.13 2.2 UBS Securities Australia Ltd Buy 2.13 3.2 UBS Securities Australia Ltd Buy 2.13 3.2 UBS Securities Australia Ltd Buy 4.154 4.30 UBS Securities Australia Ltd Buy	0-Jun-14	UBS Securities Australia Ltd	Buv	145	15	Ordinary
UBS Securities Australia Ltd Buy 7,113 7,10 UBS Securities Australia Ltd Buy 7,113 7,13 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,3 3,956 UBS Securities Australia Ltd Buy 2,3 3 UBS Securities Australia Ltd Buy 2,13 2,2 UBS Securities Australia Ltd Buy 8,56 1,000 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,16 UBS Securities Australia Ltd Buy 4	Q-1.0-1.4	I IBS Societies Australia I to	73.0	920	2 5	o diam's
UBS Securities Australia Ltd Buy 105 11 UBS Securities Australia Ltd Buy 106 11 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,33 3 UBS Securities Australia Ltd Buy 2,56 887 UBS Securities Australia Ltd Buy 2,56 1,000 UBS Securities Australia Ltd Buy 3,56 1,000 UBS Securities Australia Ltd Buy 1,565 1,60 UBS Securities Australia Ltd Buy 1,565 1,60 UBS Securities Australia Ltd Buy 1,567 4,14 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,154 4,10 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 </td <td>0.1.0</td> <td>LIPS Societies Australia 14d</td> <td>52</td> <td>0/6</td> <td>2 }</td> <td>Crairialy</td>	0.1.0	LIPS Societies Australia 14d	52	0/6	2 }	Crairialy
UBS Securities Australia Ltd Buy 106 11 UBS Securities Australia Ltd Buy 100 13 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 22 22 UBS Securities Australia Ltd Buy 8,568 857 UBS Securities Australia Ltd Buy 6,660 1,000 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154	111111111111111111111111111111111111111	Obs. Securities Abstraila Ltd	ing	611,1	/36	Ordinary
UBS Securities Australia Ltd Buy 10 UBS Securities Australia Ltd Buy 38,022 3,396 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 23 3 UBS Securities Australia Ltd Buy 213 23 UBS Securities Australia Ltd Buy 213 23 UBS Securities Australia Ltd Buy 213 34 UBS Securities Australia Ltd Buy 1,565 1,000 UBS Securities Australia Ltd Buy 1,565 1,607 UBS Securities Australia Ltd Buy 1,565 1,607 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154 420 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 4,74	0-Jun-14	UBS Securities Australia Ltd	Buy	106	11	Ordinary
UBS Securities Australia Ltd Buy 38,022 3,936 UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,13 136 UBS Securities Australia Ltd Buy 2,13 2,23 UBS Securities Australia Ltd Buy 8,568 887 UBS Securities Australia Ltd Buy 9,660 1,000 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,579 4,14 UBS Securities Australia Ltd Buy 4,579 4,14 UBS Securities Australia Ltd Buy 4,579 4,14 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy	0-Jun-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
UBS securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 2,33 3 UBS Securities Australia Ltd Buy 2,13 2.2 UBS Securities Australia Ltd Buy 8,568 887 UBS Securities Australia Ltd Buy 9,660 1,000 UBS Securities Australia Ltd Buy 1,565 1,60 UBS Securities Australia Ltd Buy 4,154 4,10 UBS Securities Australia Ltd Buy 4,154 4,00 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 1,74 UBS Securities Australia Ltd Buy <t< td=""><td>0-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>38,022</td><td>3,936</td><td>Ordinary</td></t<>	0-Jun-14	UBS Securities Australia Ltd	Buy	38,022	3,936	Ordinary
UBS Securities Australia Ltd Buy 1,314 136 UBS Securities Australia Ltd Buy 23 3 UBS Securities Australia Ltd Buy 213 22 UBS Securities Australia Ltd Buy 8,568 887 UBS Securities Australia Ltd Buy 9,650 1,000 UBS Securities Australia Ltd Buy 1,565 1,82 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 1,526	0-Jun-14	UBS Securities Australia Ltd	Buv	1.314	136	Ordinary
UBS Securities Australia Ltd Buy 23 35 UBS Securities Australia Ltd Buy 213 22 UBS Securities Australia Ltd Buy 8,568 887 UBS Securities Australia Ltd Buy 1,660 1,000 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 1,526 1,58 UBS Securities Australia Ltd Buy 1,570 1,736 UBS Securities Australia Ltd Buy 1,570 1,736 UBS Securities Australia Ltd Buy 1,57	0-Jun-14	UBS Securities Australia Ltd	Bis	1314	136	Ordin any
UBS Securities Australia Ltd Buy 2.3 2.3 UBS Securities Australia Ltd Buy 8.568 887 UBS Securities Australia Ltd Buy 9.660 1,000 UBS Securities Australia Ltd Buy 1,565 162 UBS Securities Australia Ltd Buy 1,565 162 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 1,526 173 UBS Securities Australia Ltd Buy 1,526 173 UBS Securities Australia Ltd Buy 1,5770	0.0 0-14	I IBS Sacurities Australia I td	6.00		2 (
UBS Securities Australia Ltd Buy 8.568 887 UBS Securities Australia Ltd Buy 9,660 1,000 UBS Securities Australia Ltd Buy 9,660 1,000 UBS Securities Australia Ltd Buy 1,555 162 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154 400 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 188 UBS Securities Australia Ltd Buy 1,526 188 UBS Securities Australia Ltd Buy 1,570 1,736 UBS Securities Australia Ltd Buy 1,570 1,736 UBS Securities Australia Ltd Buy <t< td=""><td></td><td>I DO Commission Assessment Land</td><td>600</td><td>67</td><td></td><td>Ciginary</td></t<>		I DO Commission Assessment Land	600	67		Ciginary
UBS Securities Australia Ltd Buy 8,568 887 UBS Securities Australia Ltd Buy 9,660 1,000 UBS Securities Australia Ltd Buy 3,28 3,4 UBS Securities Australia Ltd Buy 4,154 4,90 UBS Securities Australia Ltd Buy 4,154 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 1,526 1,58 UBS Securities Australia Ltd Buy 1,526 1,58 UBS Securities Australia Ltd Buy 1,570 1,730 UBS Securities Australia Ltd Buy 1,5770 1,730 UBS Securities Australia Ltd Buy	0-2011-14	Obs securities Australia Lt0	Buy	213	777	Ordinary
UBS Securities Australia Ltd Buy 9,660 1,000 UBS Securities Australia Ltd Buy 328 34 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,579 4/7 UBS Securities Australia Ltd Buy 565 158 UBS Securities Australia Ltd Buy 565 172 UBS Securities Australia Ltd Buy 5770 1,736 UBS Securities Australia Ltd Buy 5770 1,736 UBS Securities Australia Ltd Buy 5770<	0-Jun-14	UBS Securities Australia Ltd	Buy	8,568	887	Ordinary
UBS Securities Australia Ltd Buy 328 34 UBS Securities Australia Ltd Buy 1,565 162 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 3,999 414 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 1,526 168 UBS Securities Australia Ltd Buy 1,526 170 1,736 UBS Securities Australia Ltd Buy 985 102 102 102 UBS Securities Australia Ltd Buy 1,770 1,770 1,770 1,770 1,770 1,770 1,770 1,770 1,770 <td< td=""><td>0-Jun-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>099'6</td><td>1,000</td><td>Ordinary</td></td<>	0-Jun-14	UBS Securities Australia Ltd	Buy	099'6	1,000	Ordinary
UBS Securities Australia Ltd Buy 1,565 162 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 3,999 414 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 1,526 168 UBS Securities Australia Ltd Buy 1,570 1,736 UBS Securities Australia Ltd Buy 16,770 1,736 UBS Securities Australia Ltd Buy 16,770 1,736 UBS Securities Australia Ltd Buy 97 97	0-Jun-14	UBS Securities Australia Ltd	Buv	828	34	Ordinany
UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 1,58 UBS Securities Australia Ltd Buy 1,526 1,62 UBS Securities Australia Ltd Buy 1,526 1,62 UBS Securities Australia Ltd Buy 1,679 1,736 UBS Securities Australia Ltd Buy 1670 1,736 UBS Securities Australia Ltd Buy 1670 1,736 UBS Securities Australia Ltd Buy 1670 1,736	D-1 m-14	I IBS Securities Australia Ltd		1 565	200	
UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 3,999 4,154 430 UBS Securities Australia Ltd Buy 4,154 4,30 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,526 158 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 1,526 168 UBS Securities Australia Ltd Buy 16770 1,736 UBS Securities Australia Ltd Buy 16770 1,736 UBS Securities Australia Ltd Buy 937 937	14	LIDO CONTRIBUTO A LANGE OF THE CONTRIBUTION OF		200.1	70.	Gollaly S
UBS Securities Australia Ltd Buy 3,999 414 UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 1,526 1,58 UBS Securities Australia Ltd Buy 1,526 1,52 UBS Securities Australia Ltd Buy 1,6770 1,736 UBS Securities Australia Ltd Buy 16,770 1,736 UBS Securities Australia Ltd Buy 16,770 1,736	0-70m-14	OBS Seculities Australia Ltd	Buy	4,154	430	Ordinary
UBS Securities Australia Ltd Buy 4,154 430 UBS Securities Australia Ltd Buy 1,067 1,067 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 1,58 UBS Securities Australia Ltd Buy 167 1,73 UBS Securities Australia Ltd Buy 937 97	0-Jun-14	UBS Securities Australia Ltd	Buy	3,999	414	Ordinary
UBS Securities Australia Ltd Buy 10,307 1,067 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 985 102 UBS Securities Australia Ltd Buy 16,770 1,736 UBS Securities Australia Ltd Buy 937 97	0-Jun-14	UBS Securities Australia Ltd	Buy	4,154	430	Ordinary
UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 985 102 UBS Securities Australia Ltd Buy 16,70 1,736 UBS Securities Australia Ltd Buy 937 97	0-Jun-14	UBS Securities Australia Ltd	Buv	10.307	1 067	Ordinary
UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 1,526 102 UBS Securities Australia Ltd Buy 16770 1,736 UBS Securities Australia Ltd Buy 937 977	0-lun-14	I IRS Sequitible Australia Ltd	Busy	4 570	1774	Ordinary
UBS Securities Australia Ltd Buy 4,5/9 474 UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 985 1,02 UBS Securities Australia Ltd Buy 16,770 1,736 UBS Securities Australia Ltd Buy 937 97	1 1 1 1	COO OCCURRED FIG.	and.	4,079	4/4	Ordinary
UBS Securities Australia Ltd Buy 1,526 158 UBS Securities Australia Ltd Buy 985 102 UBS Securities Australia Ltd Buy 16770 1,736 UBS Securities Australia Ltd Buy 937 97	0-Jun-14	UBS Securities Australia Ltd	Buy	4,579	474	Ordinary
UBS Securities Australia Ltd Buy 985 102 UBS Securities Australia Ltd Buy 16,770 1,736 UBS Securities Australia Ltd Buy 937 97	J-Jun-14	UBS Securities Australia Ltd	Buy	1,526	158	Ordinary
UBS Securities Australia Ltd Buy 16,770 1,736 UBS Securities Australia Ltd Buy 937 97	J-Jun-14	UBS Securities Australia Ltd	Buy	982	102	Ordinary
UBS Securities Australia Ltd Buy 937 97	J-Jun-14	UBS Securities Australia Ltd	Buv	16.770	1,736	Ordinary
Obs Securities Australia Ltu Buy	2-1-in-14	I IBC Counties Australia I to	p	200	100	(in the C
	17	חסם שנישוונים שתפוושוש דות	Buy	700	76	Crainary

USS Securities Australia Ltd	30-Jun-14	UBS Securities Australia Ltd	Buv	969'9	200	Crotoan
USS Securities Australia Ltd Bluy 65,776 6778 67	4	UBS Securities Australia Ltd	Buv	4,552	470	Ordinary
USS Securities Australia Ltd	30-Jun-14	UBS Securities Australia Ltd	Buy	6,570	678	Ordinary
USS Securities Australia Ltd Bluy 17,122 1,105 1,055	30-Jun-14	UBS Securities Australia Ltd	Buy	63,741	6,578	Ordinary
USS Securities Australia Ltd Buy 17,772 1,767 1,067	4	UBS Securities Australia Ltd	Buy	66,158	6,831	Ordinary
UBS Securities Australia Ltd Buy 1,050 129 129 UBS Securities Australia Ltd Buy 2,210 2,219 2,219 1,29 UBS Securities Australia Ltd Buy 5,219 1,59 1,69 1,59 1,69 1,59 1,69 1,59 1,69 1,59 1,69 1,69 1,59 1,69 <td< td=""><td>4</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>17,122</td><td>1,767</td><td>Ordinary</td></td<>	4	UBS Securities Australia Ltd	Buy	17,122	1,767	Ordinary
UBS Securities Australia Ltd Buy 1802 189 UBS Securities Australia Ltd Buy 2.165 2.23 UBS Securities Australia Ltd Buy 2.165 2.23 LBS Securities Australia Ltd Buy 2.165 2.23 LBS Securities Australia Ltd Buy 2.168 2.241 LBS Securities Australia Ltd Buy 2.264 2.241 LBS Securities Australia Ltd Buy 2.254 2.251 LBS Securities Australia Ltd Buy 2.75 2.85 LBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	10,717	1,106	Ordinary
UBS Securities Australia Ltd Buy 5.1219 2.28 UBS Securities Australia Ltd Buy 5.168 2.29 LBS Securities Australia Ltd Buy 5.168 2.29 LBS Securities Australia Ltd Buy 5.254 5.239 LBS Securities Australia Ltd Buy 2.254 2.291 LBS Securities Australia Ltd Buy 5.254 2.291 LBS Securities Australia Ltd Buy 5.204 2.254 LBS Securities Australia Ltd Buy 5.204 4.25 LBS Securities Australia Ltd Buy 5.204 4.25 LBS Securities Australia Ltd Buy 3.703 3.733 LBS Securities Australia Ltd Buy 3.704 4.25 LBS Securities Australia Ltd Bu	4	UBS Securities Australia Ltd	Buy	1,802	186	Ordinary
UBS Securities Australia Ltd Buy 5,165 538 UBS Securities Australia Ltd Buy 1,357 140 LBS Securities Australia Ltd Buy 1,357 140 LBS Securities Australia Ltd Buy 2,683 2,239 LBS Securities Australia Ltd Buy 2,583 2,231 LBS Securities Australia Ltd Buy 2,583 2,231 LBS Securities Australia Ltd Buy 2,583 2,231 LBS Securities Australia Ltd Buy 2,522 2,651 LBS Securities Australia Ltd Buy 2,523 3,647 LBS Securities Australia Ltd Buy 2,652 4,75 LBS Securities Australia Ltd Buy 2,75 2,65 LBS Securities Australia Ltd Buy 2,75 2,65 LBS Securities Australia Ltd Buy 3,75 3,72 LBS Securities Australia Ltd Buy 3,75 4,75 LBS Securities Australia Ltd Buy 3,75 4,75 LBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	2,219	229	Ordinary
UBS Securities Australia Ltd Buy 155 146 UBS Securities Australia Ltd Buy 155 140 UBS Securities Australia Ltd Buy 6.904 6.904 6.705 UBS Securities Australia Ltd Buy 2.683 2.239 UBS Securities Australia Ltd Buy 2.683 2.239 UBS Securities Australia Ltd Buy 2.683 2.675 UBS Securities Australia Ltd Buy 2.584 2.675 UBS Securities Australia Ltd Buy 5.265 6.275 UBS Securities Australia Ltd Buy 5.781 3.867 UBS Securities Australia Ltd Buy 5.767 4.27 UBS Securities Australia Ltd Buy 4.77 4.75 UBS Securities Australia Ltd Buy 4.77 4.75 UBS Securities Australia Ltd Buy 4.75 4.75 UBS Securities Australia Ltd Buy 3.764 4.75 UBS Securities Australia Ltd Buy 3.764 4.75 UBS Securities Australia Ltd	4	UBS Securities Australia Ltd	Buy	5,165	533	Ordinary
UBS Securities Australia Ltd Buy 1,357 1,40 Bound of Standing Ltd Buy 64,90 7,523 1,40 UBS Securities Australia Ltd Buy 2,1523 2,329 1,359 <t< td=""><td>4</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>155</td><td>16</td><td>Ordinary</td></t<>	4	UBS Securities Australia Ltd	Buy	155	16	Ordinary
UBS Securities Australia Ltd Bluy 2,839 2,939 UBS Securities Australia Ltd Bluy 2,835 2,233 UBS Securities Australia Ltd Bluy 2,835 2,332 UBS Securities Australia Ltd Bluy 5,731 3,547 UBS Securities Australia Ltd Bluy 5,721 3,547 UBS Securities Australia Ltd Bluy 5,721 3,547 UBS Securities Australia Ltd Bluy 5,721 3,733 3,647 UBS Securities Australia Ltd Bluy 6,727 4,27 4,27 UBS Securities Australia Ltd Bluy 6,727 4,27 4,27 UBS Securities Australia Ltd Bluy 7,604 7,604 7,604 7,604 UBS Securities Australia Ltd Bluy 7,604 7,604 7,604 7,604 UBS Securities Australia Ltd Bluy 7,604 7,604 7,604 7,604 UBS Securities Australia Ltd Bluy 7,604 7,604 7,604 7,604 7,604 UBS Securities Austral	4	UBS Securities Australia Ltd	Buy	1,357	140	Ordinary
UBS Securities Australia Ltd Buy 2,683 2,239 UBS Securities Australia Ltd Buy 2,683 2,239 UBS Securities Australia Ltd Buy 2,683 2,631 2,631 UBS Securities Australia Ltd Buy 2,373 3,647 3,673 UBS Securities Australia Ltd Buy 2,373 3,671 3,673 UBS Securities Australia Ltd Buy 2,373 3,671 4,273 UBS Securities Australia Ltd Buy 2,573 3,671 4,273 UBS Securities Australia Ltd Buy 3,723 4,273 4,273 UBS Securities Australia Ltd Buy 7,604 4,273 4,273 4,273 UBS Securities Australia Ltd Buy 7,604 4,606 4,607 4,606 4,007 4,606 4,007 4,606 4,007 4,606 4,007 4,606 4,007 4,007 4,007 4,007 4,007 4,007 4,007 4,007 4,007 4,007 4,007 4,007 4,007 4,007 <td>4</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>64,904</td> <td>6,705</td> <td>Ordinary</td>	4	UBS Securities Australia Ltd	Buy	64,904	6,705	Ordinary
UBS Securities Australia Ltd Buy 21,565 2,539 UBS Securities Australia Ltd Buy 21,562 2,539 UBS Securities Australia Ltd Buy 25,241 2,647 UBS Securities Australia Ltd Buy 25,257 3,53 UBS Securities Australia Ltd Buy 3,73 3,647 UBS Securities Australia Ltd Buy 3,73 3,647 UBS Securities Australia Ltd Buy 3,73 3,647 UBS Securities Australia Ltd Buy 6,83 1,16 4,22 UBS Securities Australia Ltd Buy 6,83 1,16 4,22 UBS Securities Australia Ltd Buy 4,05 4,07 4,22 UBS Securities Australia Ltd Buy 4,05 4,07 4,07 UBS Securities Australia Ltd Buy 3,01 3,142 4,05 UBS Securities Australia Ltd Buy 3,01 3,142 4,05 4,05 4,05 UBS Securities Australia Ltd Buy 3,01 3,142 4,05 4,05 <td>4</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>2,833</td> <td>293</td> <td>Ordinary</td>	4	UBS Securities Australia Ltd	Buy	2,833	293	Ordinary
UBS Securities Australia Ltd Buy 57,581 2,556 UBS Securities Australia Ltd Buy 57,281 2,431 UBS Securities Australia Ltd Buy 57,281 2,431 UBS Securities Australia Ltd Buy 55,287 6,739 UBS Securities Australia Ltd Buy 65,278 6,739 UBS Securities Australia Ltd Buy 64,271 6,27 UBS Securities Australia Ltd Buy 6,421 6,421 6,539 UBS Securities Australia Ltd Buy 6,421 7,122 13 UBS Securities Australia Ltd Buy 7,604 1,425 4,208 UBS Securities Australia Ltd Buy 7,604 1,504 4,775 4,208 UBS Securities Australia Ltd Buy 8,100 3,1,231 3,204 4,075 UBS Securities Australia Ltd Buy 3,1,231 3,204 4,075 4,08 UBS Securities Australia Ltd Buy 3,1,231 3,204 4,075 4,08 UBS Securities Australia Ltd Bu	4	UBS Securities Australia Ltd	Buy	21,685	2,239	Ordinary
UBS Securities Australia Ltd Buy 57,781 5,667 UBS Securities Australia Ltd Buy 23,524 2,431 UBS Securities Australia Ltd Buy 55,287 6,739 UBS Securities Australia Ltd Buy 65,287 6,739 UBS Securities Australia Ltd Buy 65,287 6,739 UBS Securities Australia Ltd Buy 68,341 7,122 UBS Securities Australia Ltd Buy 6,047 7,852 UBS Securities Australia Ltd Buy 7,024 15 UBS Securities Australia Ltd Buy 7,004 15 UBS Securities Australia Ltd Buy 7,004 1,005 UBS Securities Australia Ltd Buy 2,500 1,005 UBS Securities Australia Ltd Buy 2,500 1,005 UBS Securities Australia Ltd <	30-Jun-14	UBS Securities Australia Ltd	Buy	32,532	3,359	Ordinary
UBS Securities Australia Ltd Buy 25.34 2.431 UBS Securities Australia Ltd Buy 55.251 5.739 UBS Securities Australia Ltd Buy 55.251 5.739 UBS Securities Australia Ltd Buy 3.733 6.42 UBS Securities Australia Ltd Buy 3.733 7.122 UBS Securities Australia Ltd Buy 7.62 4.22 UBS Securities Australia Ltd Buy 7.62 4.23 UBS Securities Australia Ltd Buy 7.62 4.23 UBS Securities Australia Ltd Buy 7.675 7.69 UBS Securities Australia Ltd Buy 3.703 3.703 UBS Securities Australia Ltd Buy 3.704 4.05 UBS Securities Australia Ltd Buy 3.708 3.701 3.703 UBS Securities Australia Lt	4	UBS Securities Australia Ltd	Buy	57,781	996'5	Ordinary
UBS Securities Australia Ltd Bly 55.321 3.6.47 UBS Securities Australia Ltd Bly 5.5.27 6.7.39 UBS Securities Australia Ltd Bly 5.7.33 3.86 UBS Securities Australia Ltd Bly 4.05 4.0 UBS Securities Australia Ltd Bly 4.0 4.0 UBS Securities Australia Ltd Bly 3.0 3.0 UBS Securities Australia Ltd Bly 3.0 <t< td=""><td>4</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>23,544</td><td>2,431</td><td>Ordinary</td></t<>	4	UBS Securities Australia Ltd	Buy	23,544	2,431	Ordinary
UBS Securities Australia Ltd Buy 65,267 6,739 UBS Securities Australia Ltd Buy 3,733 386 UBS Securities Australia Ltd Buy 3,733 386 UBS Securities Australia Ltd Buy 405 22 UBS Securities Australia Ltd Buy 6,21 7,72 UBS Securities Australia Ltd Buy 6,21 7,72 UBS Securities Australia Ltd Buy 7,72 7,82 UBS Securities Australia Ltd Buy 7,72 7,82 UBS Securities Australia Ltd Buy 4,75 7,82 UBS Securities Australia Ltd Buy 7,103 1,22 UBS Securities Australia Ltd Buy 7,103 1,03 UBS Securities Australia Ltd Buy 25,01 1,03 UBS Securities Australia Ltd Buy 25,01 1,03 UBS Securities Australia Ltd Buy 2,03 1,03 UBS Securities Australia Ltd Buy 2,04 1,03 UBS Securities Australia Ltd Buy 2,	4	UBS Securities Australia Ltd	Buy	35,321	3,647	Ordinary
UBS Securities Australia Ltd Buy 57.3 586 UBS Securities Australia Ltd Buy 37.31 586 UBS Securities Australia Ltd Buy 17.2 1.2 UBS Securities Australia Ltd Buy 6.421 6.63 UBS Securities Australia Ltd Buy 6.421 7.632 UBS Securities Australia Ltd Buy 4.05 7.832 UBS Securities Australia Ltd Buy 4.05 7.822 UBS Securities Australia Ltd Buy 4.05 7.822 UBS Securities Australia Ltd Buy 4.05 7.821 UBS Securities Australia Ltd Buy 4.05 4.075 UBS Securities Australia Ltd Buy 3.0,487 4.075 UBS Securities Australia Ltd Buy 4.075 4.075 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	65,267	6,739	Ordinary
UBS Securities Australia Ltd Buy 3,733 386 UBS Securities Australia Ltd Buy 406 42 UBS Securities Australia Ltd Buy 68,941 7,122 UBS Securities Australia Ltd Buy 64,21 65,64 UBS Securities Australia Ltd Buy 4,775 405 UBS Securities Australia Ltd Buy 4,775 405 UBS Securities Australia Ltd Buy 76,047 7,822 UBS Securities Australia Ltd Buy 4,775 405 UBS Securities Australia Ltd Buy 4,775 405 UBS Securities Australia Ltd Buy 3,031 3,043 UBS Securities Australia Ltd Buy 3,042 4,057 UBS Securities Australia Ltd Buy 3,043 4,05 UBS Securities Australia Ltd Buy 3,041 3,041 UBS Securities Australia Ltd Buy 3,042 4,05 UBS Securities Australia Ltd Buy 3,041 4,05 UBS Securities Australia Ltd Buy	30-Jun-14	UBS Securities Australia Ltd	Buy	- 28	9	Ordinary
UBS Securities Australia Ltd Bby 116 12 UBS Securities Australia Ltd Buy 6421 662 UBS Securities Australia Ltd Buy 6,421 663 UBS Securities Australia Ltd Buy 6,421 6,632 UBS Securities Australia Ltd Buy 4,775 493 UBS Securities Australia Ltd Buy 4,775 493 UBS Securities Australia Ltd Buy 4,775 405 UBS Securities Australia Ltd Buy 4,775 408 UBS Securities Australia Ltd Buy 4,675 4,075 UBS Securities Australia Ltd Buy 32,417 4,075 UBS Securities Australia Ltd Buy 32,417 4,075 UBS Securities Australia Ltd Buy 32,417 4,075 UBS Securities Australia Ltd Buy 3,048 4,075 UBS Securities Australia Ltd Buy 3,048 4,075 UBS Securities Australia Ltd Buy 3,048 4,075 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	3,733	386	Ordinary
UBS Securities Australia Ltd Buy 64.21 6.64 UBS Securities Australia Ltd Buy 6.421 7.624 UBS Securities Australia Ltd Buy 6.421 6.64 UBS Securities Australia Ltd Buy 6.421 6.64 UBS Securities Australia Ltd Buy 7.6,047 7.852 UBS Securities Australia Ltd Buy 4.775 4.85 UBS Securities Australia Ltd Buy 4.775 4.85 UBS Securities Australia Ltd Buy 3.1031 3.204 UBS Securities Australia Ltd Buy 3.645 4.006 UBS Securities Australia Ltd Buy 3.645 4.006 UBS Securities Australia Ltd Buy 3.645 4.006 UBS Securities Australia Ltd Buy 3.724 4.00 UBS Securities Australia Ltd Buy 2.510 2.591 UBS Securities Australia Ltd Buy 2.728 2.591 UBS Securities Australia Ltd Buy 2.728 2.756 UBS Securities Australia Ltd	4	UBS Securities Australia Ltd	Buv	116	12	Ordinary
UBS Securities Australia Ltd Bby 68,941 7,122 UBS Securities Australia Ltd Buy 6,421 664 UBS Securities Australia Ltd Buy 76,047 7852 UBS Securities Australia Ltd Buy 4,775 493 UBS Securities Australia Ltd Buy 4,775 499 UBS Securities Australia Ltd Buy 4,775 4,809 UBS Securities Australia Ltd Buy 31,041 1,45 UBS Securities Australia Ltd Buy 31,048 3,264 UBS Securities Australia Ltd Buy 31,048 3,243 UBS Securities Australia Ltd Buy 31,048 3,243 UBS Securities Australia Ltd Buy 37,238 5,510 UBS Securities Australia Ltd Buy 37,238 5,510 UBS Securities Australia Ltd Buy 5,109 3,243 UBS Securities Australia Ltd Buy 5,210 2,285 UBS Securities Australia Ltd Buy 3,245 1,486 UBS Securities Australia Ltd <	4	UBS Securities Australia Ltd	Buv	406	42	Ordinary
UBS Securities Australia Ltd Buy 6,421 664 UBS Securities Australia Ltd Buy 76,047 7,852 UBS Securities Australia Ltd Buy 4,775 4,33 UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 31,031 3,044 UBS Securities Australia Ltd Buy 31,031 3,044 UBS Securities Australia Ltd Buy 25,617 2,645 UBS Securities Australia Ltd Buy 31,031 3,031 UBS Securities Australia Ltd Buy 25,617 2,645 UBS Securities Australia Ltd Buy 37,238 5,910 UBS Securities Australia Ltd Buy 37,238 5,910 UBS Securities Australia Ltd Buy 37,238 3,412 UBS Securities Australia Ltd Buy 3,756 3,64 UBS Securities Australia Ltd Buy 2,756 2,85 UBS Securities Australia Ltd Buy 2,756 2,85 UBS Securities Australia Ltd <td< td=""><td>4</td><td>UBS Securities Australia Ltd</td><td>BUV</td><td>68.941</td><td>7.122</td><td>Ordinary</td></td<>	4	UBS Securities Australia Ltd	BUV	68.941	7.122	Ordinary
UBS Securities Australia Ltd Buy 76,047 7,852 UBS Securities Australia Ltd Buy 4,775 493 UBS Securities Australia Ltd Buy 4,775 149 UBS Securities Australia Ltd Buy 4,06 4,2 UBS Securities Australia Ltd Buy 31,031 3,204 UBS Securities Australia Ltd Buy 36,61 4,808 UBS Securities Australia Ltd Buy 36,61 2,610 UBS Securities Australia Ltd Buy 36,73 3,043 UBS Securities Australia Ltd Buy 37,238 3,043 UBS Securities Australia Ltd Buy 37,238 3,043 UBS Securities Australia Ltd Buy 37,238 3,043 UBS Securities Australia Ltd Buy 27,238 3,043 UBS Securities Australia Ltd Buy 2,756 2,85 UBS Securities Australia Ltd Buy 2,756 2,85 UBS Securities Australia Ltd Buy 2,756 2,86 UBS Securities Australia Ltd B	4	UBS Securities Australia Ltd	Boy	6.421	664	Ordinary
UBS Securities Australia Ltd Buy 4,775 499 UBS Securities Australia Ltd Buy 4,775 4,22 23 UBS Securities Australia Ltd Buy 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 406 42 480	14	UBS Securities Australia Ltd	Buv	76.047	7.852	Ordinary
UBS Securities Australia Ltd Buy 222 23 UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 31,031 3,044 UBS Securities Australia Ltd Buy 25,617 2,645 UBS Securities Australia Ltd Buy 35,738 3,103 UBS Securities Australia Ltd Buy 25,617 2,645 UBS Securities Australia Ltd Buy 33,410 3,412 UBS Securities Australia Ltd Buy 33,610 3,243 UBS Securities Australia Ltd Buy 33,610 3,243 UBS Securities Australia Ltd Buy 3,517 5,118 UBS Securities Australia Ltd Buy 3,600 3,600 UBS Securities Australia Ltd Buy 2,756 2,851 UBS Securities Australia Ltd Buy 2,756 2,851 UBS Securities Australia Ltd Buy 2,756 2,851 UBS Securities Australia Ltd Buy </td <td>4</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>4 775</td> <td>763</td> <td>Ordinary</td>	4	UBS Securities Australia Ltd	Buv	4 775	763	Ordinary
UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 31,031 3,204 UBS Securities Australia Ltd Buy 31,031 3,204 UBS Securities Australia Ltd Buy 25,617 2,645 UBS Securities Australia Ltd Buy 33,437 3,938 UBS Securities Australia Ltd Buy 25,107 2,617 UBS Securities Australia Ltd Buy 37,238 3,243 UBS Securities Australia Ltd Buy 37,238 3,018 UBS Securities Australia Ltd Buy 37,238 3,018 UBS Securities Australia Ltd Buy 37,238 3,018 UBS Securities Australia Ltd Buy 4,446 4,05 UBS Securities Australia Ltd Buy 2,756 2,85 UBS Securities Australia Ltd <t< td=""><td>4</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>222</td><td>23</td><td>Ordinary</td></t<>	4	UBS Securities Australia Ltd	Buy	222	23	Ordinary
UBS Securities Australia Ltd Buy 406 42 UBS Securities Australia Ltd Buy 31,031 3104 UBS Securities Australia Ltd Buy 46,565 4,808 UBS Securities Australia Ltd Buy 2,5617 2,645 UBS Securities Australia Ltd Buy 35,617 2,645 UBS Securities Australia Ltd Buy 37,238 3,048 UBS Securities Australia Ltd Buy 37,238 3,011 UBS Securities Australia Ltd Buy 37,238 3,01 UBS Securities Australia Ltd Buy 37,238 3,01 UBS Securities Australia Ltd Buy 3,04 4,07 UBS Securities Australia Ltd Buy 3,04 4,07 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,214 2,29 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,214 2,29 UBS Securities Australia Ltd Buy <td>4</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>145</td> <td>15</td> <td>Ordinary</td>	4	UBS Securities Australia Ltd	Buy	145	15	Ordinary
UBS Securities Australia Ltd Buy 31,031 3,044 UBS Securities Australia Ltd Buy 31,031 3,044 UBS Securities Australia Ltd Buy 25,617 2,645 UBS Securities Australia Ltd Buy 25,617 2,645 UBS Securities Australia Ltd Buy 25,107 2,591 UBS Securities Australia Ltd Buy 33,011 3,412 UBS Securities Australia Ltd Buy 33,011 3,412 UBS Securities Australia Ltd Buy 33,011 3,412 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,756 2,85 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,756 2,85 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd <td< td=""><td>4</td><td>UBS Securities Australia Ltd</td><td>Buv</td><td>406</td><td>42</td><td>Ordinary</td></td<>	4	UBS Securities Australia Ltd	Buv	406	42	Ordinary
UBS Securities Australia Ltd Buy 31,031 3,204 UBS Securities Australia Ltd Buy 48,565 4,808 UBS Securities Australia Ltd Buy 38,430 3,968 UBS Securities Australia Ltd Buy 38,437 3,968 UBS Securities Australia Ltd Buy 25,107 2,510 UBS Securities Australia Ltd Buy 31,408 3,243 UBS Securities Australia Ltd Buy 31,408 3,243 UBS Securities Australia Ltd Buy 49,517 5,118 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,766 2,767 UBS Securities Australia Ltd Buy 2,766 7,78 UBS Securities Australia Ltd	14	UBS Securities Australia Ltd	Buy	145	15	Ordinary
UBS Securities Australia Ltd Buy 46.565 4.808 UBS Securities Australia Ltd Buy 25.617 2.645 UBS Securities Australia Ltd Buy 36.70 2.591 UBS Securities Australia Ltd Buy 25.107 2.591 UBS Securities Australia Ltd Buy 37.238 5.910 UBS Securities Australia Ltd Buy 37.038 5.01 UBS Securities Australia Ltd Buy 49.517 5.18 UBS Securities Australia Ltd Buy 49.517 5.18 UBS Securities Australia Ltd Buy 4.446 460 UBS Securities Australia Ltd Buy 2.756 2.85 UBS Securities Australia Ltd Buy 2.756 2.85 UBS Securities Australia Ltd Buy 2.756 2.85 UBS Securities Australia Ltd Buy 2.744 4.75 UBS Securities Australia Ltd Buy 2.746 4.75 UBS Securities Australia Ltd Buy 4.758 6.71 UBS Securities Australia Ltd B	14	UBS Securities Australia Ltd	Buy	31,031	3,204	Ordinary
UBS Securities Australia Ltd Buy 25,617 2,645 UBS Securities Australia Ltd Buy 32,813 3,588 UBS Securities Australia Ltd Buy 25,107 2,591 UBS Securities Australia Ltd Buy 25,107 2,591 UBS Securities Australia Ltd Buy 31,018 3,412 UBS Securities Australia Ltd Buy 33,011 3,412 UBS Securities Australia Ltd Buy 8,517 5,118 UBS Securities Australia Ltd Buy 8,517 5,118 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,756 2,85 UBS Securities Australia Ltd Buy 2,756 2,89 UBS Securities Australia Ltd Buy 7,610 1,41 UBS Securities Australia Ltd <td< td=""><td>4</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>46,565</td><td>4,808</td><td>Ordinary</td></td<>	4	UBS Securities Australia Ltd	Buy	46,565	4,808	Ordinary
UBS Securities Australia Ltd Buy 38,430 3,968 UBS Securities Australia Ltd Buy 25,107 2,531 UBS Securities Australia Ltd Buy 27,238 5,910 UBS Securities Australia Ltd Buy 31,408 3,243 UBS Securities Australia Ltd Buy 49,517 5,118 UBS Securities Australia Ltd Buy 43,768 16 UBS Securities Australia Ltd Buy 2,766 285 UBS Securities Australia Ltd Buy 1,571 141 UBS Securities Australia Ltd Buy 1,579 141 UBS Securities Australia Ltd Buy 1,579 1,41 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	25,617	2,645	Ordinary
UBS Securities Australia Ltd Buy 23,487 4,075 UBS Securities Australia Ltd Buy 25,310 2,591 UBS Securities Australia Ltd Buy 31,408 5,243 UBS Securities Australia Ltd Buy 31,408 3,243 UBS Securities Australia Ltd Buy 44,517 5,118 UBS Securities Australia Ltd Buy 44,446 460 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 4,579 4,47 UBS Securities Australia Ltd Buy 7,501 286 UBS Securities Australia Ltd Buy 7,501 286 UBS Securities Australia Ltd Buy 7,601 7,601 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	38,430	3,968	Ordinary
UBS Securities Australia Ltd Buy 25,107 2,591 UBS Securities Australia Ltd Buy 31,408 3,412 UBS Securities Australia Ltd Buy 31,408 3,412 UBS Securities Australia Ltd Buy 33,011 3,412 UBS Securities Australia Ltd Buy 8,517 5,118 UBS Securities Australia Ltd Buy 4,446 400 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 6,753 6,71 UBS Securities Australia Ltd Buy 7,601 7,80 UBS Securities Australia Ltd Buy 7,601 7,80 UBS Securities Australia Ltd Buy 7,601 7,80 UBS Securities Australia Ltd Buy	14	UBS Securities Australia Ltd	Buy	39,487	4,075	Ordinary
UBS Securities Australia Ltd Buy 57,238 5,910 UBS Securities Australia Ltd Buy 31,408 3,243 UBS Securities Australia Ltd Buy 33,011 3,412 UBS Securities Australia Ltd Buy 45,517 5,118 UBS Securities Australia Ltd Buy 4,445 460 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 245 UBS Securities Australia Ltd Buy 2,756 245 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 6,353 6,10 UBS Securities Australia Ltd Buy 6,353 6,10 UBS Securities Australia Ltd Buy 6,353 6,01 UBS Securities Australia Ltd Buy 6,950 7,18 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buv	25.107	2.591	Ordinary
UBS Securities Australia Ltd Buy 31,408 3,243 UBS Securities Australia Ltd Buy 48,517 5,118 UBS Securities Australia Ltd Buy 4,517 5,118 UBS Securities Australia Ltd Buy 1,335 138 UBS Securities Australia Ltd Buy 4,445 460 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 4,579 4,14 UBS Securities Australia Ltd Buy 4,579 4,14 UBS Securities Australia Ltd Buy 6,533 657 UBS Securities Australia Ltd Buy 6,533 610 UBS Securities Australia Ltd Buy 6,533 610 UBS Securities Australia Ltd Buy 6,950 7,180 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	57,238	5.910	Ordinary
UBS Securities Australia Ltd Buy 33,011 3,412 UBS Securities Australia Ltd Buy 49,517 5,118 UBS Securities Australia Ltd Buy 8,708 900 UBS Securities Australia Ltd Buy 4,446 400 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,714 2,291 247 UBS Securities Australia Ltd Buy 4,579 4,74 465 465 UBS Securities Australia Ltd Buy 7,601 7,86 1,341 141 141 141 142	4	UBS Securities Australia Ltd	Buv	31,408	3.243	Ordinary
UBS Securities Australia Ltd Buy 49,517 5,118 UBS Securities Australia Ltd Buy 8,708 900 UBS Securities Australia Ltd Buy 1,335 16 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 245 UBS Securities Australia Ltd Buy 4,579 4,74 UBS Securities Australia Ltd Buy 6,353 6,01 UBS Securities Australia Ltd Buy 6,353 6,01 UBS Securities Australia Ltd Buy 6,363 6,01 UBS Securities Australia Ltd Buy 6,98 7,601 7,80 UBS Securities Australia Ltd Buy 6,91 7,18 UBS Securities Australia Ltd Buy 6,91 7,18 UBS Securities Australia Ltd <t< td=""><td>4</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>33.011</td><td>3,412</td><td>Ordinary</td></t<>	4	UBS Securities Australia Ltd	Buy	33.011	3,412	Ordinary
UBS Securities Australia Ltd Buy 8,708 900 UBS Securities Australia Ltd Buy 1,335 138 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 245 UBS Securities Australia Ltd Buy 2,756 245 UBS Securities Australia Ltd Buy 2,756 245 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 6,533 657 UBS Securities Australia Ltd Buy 6,353 650 UBS Securities Australia Ltd Buy 6,853 610 UBS Securities Australia Ltd Buy 6,950 7,180 UBS Securities Australia Ltd Buy 6,914 7,160 UBS Securities Australia Ltd Buy 6,914 7,180 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	49.517	5.118	Ordinary
UBS Securities Australia Ltd Buy 1,335 138 UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,99 31 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,714 2,234 UBS Securities Australia Ltd Buy 1,541 1,41 UBS Securities Australia Ltd Buy 1,363 1,41 UBS Securities Australia Ltd Buy 7,601 7,601 UBS Securities Australia Ltd Buy 7,601 7,601 UBS Securities Australia Ltd Buy 7,601 7,601 UBS Securities Australia Ltd Buy 6,959 6,10 UBS Securities Australia Ltd Buy 6,950 7,180 UBS Securities Australia Ltd Buy 6,950 7,180 UBS Securities Australia Ltd Buy 6,914 7,15 UBS Securities Australia Ltd Buy 6,914 7,15 UBS Securities Australia Ltd Buy	4	UBS Securities Australia Ltd	Buy	8.708	006	Ordinary
UBS Securities Australia Ltd Buy 4.446 460 UBS Securities Australia Ltd Buy 2.95 31 UBS Securities Australia Ltd Buy 2.756 285 UBS Securities Australia Ltd Buy 2.756 285 UBS Securities Australia Ltd Buy 2.756 285 UBS Securities Australia Ltd Buy 4.579 474 UBS Securities Australia Ltd Buy 4.579 474 UBS Securities Australia Ltd Buy 6.353 657 UBS Securities Australia Ltd Buy 6.363 600 UBS Securities Australia Ltd Buy 6.96 7.2 UBS Securities Australia Ltd Buy 6.514 7.15 UBS Securities Australia Ltd Buy 6.514 7.15 UBS Securities Australia Ltd Buy 6.514	4	UBS Securities Australia Ltd	Buv	1.335	138	Ordinary
UBS Securities Australia Ltd Buy 4,446 460 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 245 UBS Securities Australia Ltd Buy 2,214 229 UBS Securities Australia Ltd Buy 1,4886 1,541 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 6,533 657 UBS Securities Australia Ltd Buy 5,899 610 UBS Securities Australia Ltd Buy 6,893 610 UBS Securities Australia Ltd Buy 6,893 610 UBS Securities Australia Ltd Buy 6,995 7,180 UBS Securities Australia Ltd Buy 6,914 7,150 UBS Securities Australia Ltd Buy 6,914 7,150 UBS Securities Australia Ltd Buy 6,914 7,180 UBS Securities Australia Ltd Buy 6,914 7,150 UBS Securities Australia Ltd Buy	4	UBS Securities Australia 11d	Buv	155	16	Ordinary
UBS Securities Australia Ltd Buy 299 31 UBS Securities Australia Ltd Buy 296 31 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,714 229 UBS Securities Australia Ltd Buy 2,714 229 UBS Securities Australia Ltd Buy 4,579 1,41 UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 5,839 610 UBS Securities Australia Ltd Buy 6,950 1,1 UBS Securities Australia Ltd Buy 6,950 7,1 UBS Securities Australia Ltd Buy 6,950 7,1 UBS Securities Australia Ltd Buy 6,950 7,1 UBS Securities Australia Ltd Buy 6,914 7,1 UBS Securities Australia Ltd Buy 6,914 7,1 UBS Securities Australia Ltd Buy 6,914		1 BC Counting Australia 14d	B. 05	0000	0.94	Ordinary
UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,756 285 UBS Securities Australia Ltd Buy 2,714 2,229 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 6,579 474 UBS Securities Australia Ltd Buy 6,353 657 UBS Securities Australia Ltd Buy 5,899 610 UBS Securities Australia Ltd Buy 6,98 7,601 786 UBS Securities Australia Ltd Buy 6,98 7,7 15 UBS Securities Australia Ltd Buy 6,98 7,7 UBS Securities Australia Ltd Buy 6,98 7,1 UBS Securities Australia Ltd Buy 6,914 7,15 UBS Securities Australia Ltd Buy 6,514 7,15 UBS Securities Australia Ltd Buy 6,514 7,15 UBS Securities Australia Ltd Buy 6,514 7,15 UBS Securities Australia Ltd		PIPO Committee Australia 144	S C C C C C C C C C C C C C C C C C C C	ouc.	100	Ordinary
UBS Securities Australia Ltd Buy 2,214 229 UBS Securities Australia Ltd Buy 2,214 229 UBS Securities Australia Ltd Buy 1,4886 1,541 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 7,601 766 UBS Securities Australia Ltd Buy 5,899 610 UBS Securities Australia Ltd Buy 5,899 610 UBS Securities Australia Ltd Buy 696 72 UBS Securities Australia Ltd Buy 695 71 UBS Securities Australia Ltd Buy 695,02 71,80 UBS Securities Australia Ltd Buy 695,02 71,80 UBS Securities Australia Ltd Buy 69,10 71,80 UBS Securities Australia Ltd Buy 69,10 71,80 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy <td< td=""><td></td><td>I BS Societies Australia to</td><td>Big</td><td>2 766</td><td>2 20</td><td>Ordinary</td></td<>		I BS Societies Australia to	Big	2 766	2 20	Ordinary
UBS Securities Australia Ltd Buy 2,203 UBS Securities Australia Ltd Buy 2,214 2,29 UBS Securities Australia Ltd Buy 4,579 4,44 UBS Securities Australia Ltd Buy 6,353 657 UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 6,829 610 UBS Securities Australia Ltd Buy 6,92 7,2 UBS Securities Australia Ltd Buy 6,94 7,2 UBS Securities Australia Ltd Buy 6,94 7,1 UBS Securities Australia Ltd Buy 6,94 7,180 UBS Securities Australia Ltd Buy 6,914 7,15 UBS Securities Australia Ltd Buy 6,914 <	: 5	I IDS Cocurities Australia Ltd	Sauga Para	2,730	207	Ordinary
UBS Securities Australia Ltd Buy 14,886 1,541 UBS Securities Australia Ltd Buy 14,886 1,541 UBS Securities Australia Ltd Buy 4,579 474 UBS Securities Australia Ltd Buy 6,353 657 UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 6,96 72 UBS Securities Australia Ltd Buy 6,96 72 UBS Securities Australia Ltd Buy 6,914 7,180 UBS Securities Australia Ltd Buy 6,514 7,187 UBS Securities Australia Ltd Buy	<u>.</u> :	Obs securities Australia Ltd	óng	2,309	245	Ordinary
UBS Securities Australia Ltd Buy 4,586 1,341 UBS Securities Australia Ltd Buy 4,579 4,17 UBS Securities Australia Ltd Buy 6,353 657 UBS Securities Australia Ltd Buy 7,601 7,601 UBS Securities Australia Ltd Buy 5,839 610 UBS Securities Australia Ltd Buy 6,99 15 UBS Securities Australia Ltd Buy 69,502 7,180 UBS Securities Australia Ltd Buy 69,502 7,180 UBS Securities Australia Ltd Buy 6,914 7,150 UBS Securities Australia Ltd Buy 6,914 7,180 UBS Securities Australia Ltd Buy 6,914 7,15 UBS Securities Australia Ltd Buy </td <td>!</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>417,2</td> <td>677</td> <td>Ordinary</td>	!	UBS Securities Australia Ltd	Buy	417,2	677	Ordinary
UBS Securities Australia Ltd Buy 4.579 4.44 UBS Securities Australia Ltd Buy 1,579 141 UBS Securities Australia Ltd Buy 6,553 657 UBS Securities Australia Ltd Buy 6,589 610 UBS Securities Australia Ltd Buy 6,96 72 UBS Securities Australia Ltd Buy 6,96 72 UBS Securities Australia Ltd Buy 6,91 7,180 UBS Securities Australia Ltd Buy 6,914 7,150 UBS Securities Australia Ltd Buy 6,719 6,731 UBS Securities Australia Ltd Buy 6,710 6,731 UBS Securities Australia Ltd Buy 6,710 6,731 UBS Securities Australia Ltd Buy	!	UBS Securities Australia Lto	Buy	14,886	1,541	Ordinary
UBS Securities Australia Ltd Buy 1,363 141 UBS Securities Australia Ltd Buy 6,353 657 UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 696 72 UBS Securities Australia Ltd Buy 696 72 UBS Securities Australia Ltd Buy 695 72 UBS Securities Australia Ltd Buy 695 71 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 4,167 4,167 4,167	4	UBS Securities Australia Ltd	Buy	4,579	474	Ordinary
UBS Securities Australia Ltd Buy 6575 UBS Securities Australia Ltd Buy 7 601 766 UBS Securities Australia Ltd Buy 7 601 766 UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 696 72 UBS Securities Australia Ltd Buy 69502 7180 UBS Securities Australia Ltd Buy 69502 7180 UBS Securities Australia Ltd Buy 6514 715 UBS Securities Australia Ltd Buy 6514 715 UBS Securities Australia Ltd Buy 6514 7715 UBS Securities Australia Ltd Buy 6,314 775 UBS Securities Australia Ltd Buy 4,167 430	4	UBS Securities Australia Ltd	Buy	1,363	141	Ordinary
UBS Securities Australia Ltd Buy 7,601 786 UBS Securities Australia Ltd Buy 5,839 610 UBS Securities Australia Ltd Buy 145 72 UBS Securities Australia Ltd Buy 636 72 UBS Securities Australia Ltd Buy 433 51 UBS Securities Australia Ltd Buy 6,914 7180 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,719 6,731 UBS Securities Australia Ltd Buy 4,167 430	4	UBS Securities Australia Ltd	Buy	6,353	657	Ordinary
UBS Securities Australia Ltd Buy 5.899 610 UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 696 72 UBS Securities Australia Ltd Buy 696 72 UBS Securities Australia Ltd Buy 69502 7,180 UBS Securities Australia Ltd Buy 6,514 7,180 UBS Securities Australia Ltd Buy 6,514 7,180 UBS Securities Australia Ltd Buy 6,514 7,15 UBS Securities Australia Ltd Buy 6,514 7,15 UBS Securities Australia Ltd Buy 6,514 7,15 UBS Securities Australia Ltd Buy 6,731 8,731 UBS Securities Australia Ltd Buy 4,167 4,167 4,107	4	UBS Securities Australia Ltd	Buy	1,601	786	Ordinary
UBS Securities Australia Ltd Buy 145 15 UBS Securities Australia Ltd Buy 696 72 UBS Securities Australia Ltd Buy 116 72 UBS Securities Australia Ltd Buy 69,502 7,180 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,714 715 UBS Securities Australia Ltd Buy 4,167 430	4	UBS Securities Australia Ltd	Buy	5,899	610	Ordinary
UBS Securities Australia Ltd Buy 696 72 UBS Securities Australia Ltd Buy 116 12 UBS Securities Australia Ltd Buy 69,502 7,180 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,719 6,731 UBS Securities Australia Ltd Buy 4,167 490	4	UBS Securities Australia Ltd	Buy	145	15	Ordinary
UBS Securities Australia Ltd Buy 116 12 UBS Securities Australia Ltd Buy 493 51 UBS Securities Australia Ltd Buy 6,914 7180 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,719 6,731 UBS Securities Australia Ltd Buy 3,304 341 UBS Securities Australia Ltd Buy 4,167 430	4	UBS Securities Australia Ltd	Buy	969	72	Ordinary
UBS Securities Australia Ltd Buy 493 51 UBS Securities Australia Ltd Buy 69,502 7,180 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 65,190 6,731 UBS Securities Australia Ltd Buy 3,304 341 UBS Securities Australia Ltd Buy 4,167 430	4	UBS Securities Australia Ltd	Buy	116	12	Ordinary
UBS Securities Australia Ltd Buy 69,502 7,180 UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,5190 6,731 UBS Securities Australia Ltd Buy 3,304 341 UBS Securities Australia Ltd Buy 4,167 430	4	Securities	Buy	493	51.	Ordinary
UBS Securities Australia Ltd Buy 6,914 715 UBS Securities Australia Ltd Buy 6,719 6,731 UBS Securities Australia Ltd Buy 3,304 341 UBS Securities Australia Ltd Buy 4,167 430	4	UBS Securities Australia Ltd	Buv	69,502	7.180	Ordinary
UBS Securities Australia Ltd Buy 65,190 6,731 UBS Securities Australia Ltd Buy 3,304 341 UBS Securities Australia Ltd Buy 4,167 430	4	UBS Securities Australia Ltd	Bey	6.914	715	Ordinary
UBS Securities Australia Ltd		LIRS Securities Australia 14d	200	65 190	6 731	Ordinary Ordinary
UBS Securities Australia Ltd Buy 4,167 430	, ,	LIDS Committee Australia 14d	500	061,00	0,73	Ordinary
UBS Securities Australia Ltd Buy 4,167 430	,	UBS Securities Australia Ltd	Buy	3,304	34	Crainary
	7					

UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	4,171	430	Ordinary
JBS Securities Australia Ltd	Buy	3,153	325	Ordinary
JBS Securities Australia Ltd	Buy	1,523	157	Ordinary
JBS Securities Australia Ltd	Buy	099	89	Ordinary
UBS Securities Australia Ltd	Buy	49	25	Ordinary
JBS Securities Australia Ltd	Buy	718	74	Ordinary
urities Australia Ltd	Buy	3,153	325	Ordinary
JBS Securities Australia Ltd	Buy	17,518	1,806	Ordinary
UBS Securities Australia Ltd	Buy	19,284	1,988	Ordinary
JBS Securities Australia Ltd	Buy	4,171	430	Ordinary
UBS Securities Australia Ltd	Buy	30,962	3,192	Ordinary
UBS Securities Australia Ltd	Buy	7,954	820	Ordinary
curities Australia Ltd	Buy	204	21	Ordinary
UBS Securities Australia Ltd	Buy	4,899	505	Ordinary
Securities Australia Ltd	Buy	5.228	539	Ordinary
curities Australia Ltd	Suv	213	77	Ordinary
UBS Securities Australia I td	810	10.408	1 073	Ordinan
Curities Australia Ital	600	200,00	0,0,1	Clelliary
I IDC Commission Australia Ltd	, and	,00°,1	2 3	Clumary
contres Australia Ltd	Buy	024	8	Ordinary
ecurities Australia Ltd	Buy	52,884	5,452	Ordinary
UBS Securities Australia Ltd	Buy	7,682	792	Ordinary
JBS Securities Australia Ltd	Buy	1,911	197	Ordinary
JBS Securities Australia Ltd	Buy	213	22	Ordinary
UBS Securities Australia Ltd	Buy	23,552	2,428	Ordinary
ecurities Australia Ltd	Bov	5.199	536	Ordinary
UBS Securities Australia Ltd	Buy	4 850	Q C	Ording
UBS Securities Australia Ltd	Siz	02	-	Ordinary
IRS Sequities Australia 14d	Birry	000	100	O Gallany
ourities Australia Itd	200	070	000	Guiraly
Curities Australia Ital	, and	0/6	001	Quillary
Data interpretation of the Data	600	1,4,4,7	9	Genaly Genaly
Contries Australia Ltd	l and	15,704	610'	Ordinary
Occupies Australia Etc	Sing.	140	2 3	Ordinary
Societies Assessed Etc	ing.	040	00	Orginary
IDS Societies Australia Ltd	buy Buy	20,100	5,584	Ordinary
I DC Con without Australia Ltd	900	5557	0 0	Crainary
DC Complete Australia Ltd	ing.	4,000	101	Ordinary
ecolities Australia Lto	, and	4,540	468	Ordinary
ecurities Australia Ltd	Buy	3,327	343	Ordinary
JBS Securities Australia Ltd	Buy	208	73	Ordinary
JBS Securities Australia Ltd	Buy	31,544	3,252	Ordinary
ecurities Australia Ltd	Buy	5.206	537	Ordinary
UBS Securities Australia Ltd	Buy	233	24	Ordinan
IRC Securities Australia 14d	De la	200	1770	Olukiaiy
econices Australia Ltd	Son o	2,044	070	Orginary
UBS Securities Australia Ltd	Buy	28/	80	Ordinary
ecurities Australia Ltd	Buy	3,521	363	Ordinary
UBS Securities Australia Ltd	Buy	2,706	279	Ordinary
UBS Securities Australia Ltd	Buy	5,781	296	Ordinary
ecurities Australia Ltd	Buy	8,643	168	Ordinary
Securities Australia Ltd	Buy	29	3	Ordinary
UBS Securities Australia Ltd	Buy	7 799	808	Ordinary
Politips Australia 314	Bus	11 553	101	Colonia
IBS Securities Australia 11d	800	370 6	217	Ordings
on without Australia 144		0.00	1 2	500
I DO Commission A control and	, and	D++	n i	Cramary
Curries Australia Ltd	hay	10,1/5	1,049	Ordinary
curities Australia Ltd	Buy	698	72	Ordinary
curities Australia Ltd	Buy	2,425	250	Ordinary
UBS Securities Australia Ltd	Buy	1,882	194	Ordinary
Securities Australia Ltd	Buy	11,465	1,182	Ordinary
UBS Securities Australia Ltd	Buv	2.454	253	Ordinary
Securities Australia Ltd	Birv	407	CV	Ordinan
Curities Australia I to	8.1%	040	77	Ordinary
DC Comision American	600	678	06.0	Crumary
curties Australia Ltd	Buy	100,802	10,392	Ordinary
UBS Securities Australia Ltd	Buy	35,114	3,620	Ordinary
curities Australia Ltd	Buy	892	92	Ordinan
100 Camerina A			- 7/	, e e
OT BIRLING ALIEN	Buv	1 620	751	Ordinary

860, 860, 860, 860, 860, 860, 860, 860,	13,229 967 32,520 9,225 2,727 2,727 7,197 7,197 4,15	1,368 100 3,363	Ordinary Ordinary Ordinary Ordinary Ordinary
8 by 8 by	32,520 9,725 9,725 2,534 2,727 7,1049 1,934 7,197	3,363	Ordinary Ordinary Ordinary
8 8 9 8 9 8 9 8 9 8 9 9 8 9 9 9 9 9 9 9	32,520 9,25 2,534 2,727 7,049 1,934 7,197 4,15	3,363	Ordinary Ordinary Ordinary
Buy Buy	9,225 2,534 2,527 2,227 7,049 7,197 7,197 4,15	954	Ordinary
8 8 9 8 9 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9	2,534 2,727 7,049 1,934 7,197 4,157		Ordinary
84,6 84,7 84,7 84,7 84,7 84,7 84,7 84,7 84,7	2,727 7,049 1,934 7,197 415	262	
8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by	7,049 1,934 7,197 415	282	Ordinary
8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy 8 Buy	1,934 7,197 415 4 731	779	Ordinan
8 Buy 8 Buy	415	000	Ordinary
8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	415	745	Ordinary
Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	415		, i
800 800 800 800 800 800 800 800 800 800	7 7	£4.	Ordinary
Buy Buy Buy Buy Buy Buy Buy Buy	- C	430	Ordinary
Buy Buy Buy Buy Buy Buy Buy	6,240	P46	Ordinary
Buy Buy Buy Buy Buy	39	4	Ordinary
Buy Buy Buy Buy Buy	116	12	Ordinary
Buy Buy Buy Buy	28,072	2,906	Ordinary
Buy Buy Buy Buy	4,579	474	Ordinary
Buy Buy Buy	7 946	305	Ordinary
Buy	13,501	1 408	Ordinan
Buy	100,00	P.	o deliaiy
Buy	57/	5)	Ordinary
	618	25	Ordinary
Ang	4,579	474	Ordinary
Buv	4 154	430	Ordinary
96	791	1,	
Ang d	104	/	Ordinary
Buy	4,5/9	4/4	Ordinary
Buy	10,356	1,072	Ordinary
Buy	5,786	299	Ordinary
Buy	66.539	6 881	Ordinary
Riv	900 89	7 020	Ordinary
600	000,00	200,7	old line
pro	900,00	3,0/5	Ordinary
Buy	30,463	3,147	Ordinary
Buy	5,944	614	Ordinary
Buy	89	7	Ordinary
Buy	28,861	2,980	Ordinary
Buy	17,569	1,815	Ordinary
Buy	53,772	5,555	Ordinary
Buy	18,818	1,944	Ordinary
Buv	610	69	Ordinary
Birk	362	75	Ordinato
200	3 834	100	O'HILLIAN O
, and	3,024	CSC.	Garnary
eny	4,162	430	Ordinary
Buy	32,389	3,346	Ordinary
Buy	5,343	252	Ordinary
Buv	5.343	552	Ordinary
, m	906 63	C 412	Ordings,
λής.	056,20	0,410	Ordinary
Buy	53,/34	1,551	Ordinary
Buy	4,162	430	Ordinary
Buy	49,571	5,121	Ordinary
Buy	55.136	2.690	Ordinary
Bits	89		vacaion
,,,,	645	. 9	Ordinary
óno	040	R	Clumary
Buy	3,153	325	Ordinary
Buy	1,057	109	Ordinary
Buy	18,896	1,948	Ordinary
Buv	213	22	Ordinary
Buy	5,354	552	Ordinary
	2,047	211	digital
600	ccc		o de la composito de la compos
, and	663	17	o delialy
gnà	666	501	Ordinary
Buy	815	84	Ordinary
Buy	12,338	1,272	Ordinary
Buy	28	9	Ordinary
Buy	3.153	325	Ordinary
62	0000	250	Codinary
Buy	2,668	5/2	Ordinary
Buy	272	28	Ordinary
, and	1	3	Clourary
	8 8 9 8 9 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9	13, 13, 13, 13, 13, 13, 13, 13, 13, 13,	13,946 13,601 14,519 4,154 4,154 4,154 4,154 4,154 4,154 4,154 4,154 6,578 6,68,006 7,78 6,80 8,80 1,7,569 1,17

41-IDF-L0	COO Securities Australia Ltd	à			
01-Jul-14	UBS Securities Australia Ltd	Buy	163,917	16,945	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	231,959	23,990	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	578,537	29,880	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Sell	73,492	(009'2)	Ordinary
01-Jul-14	UBS Wealth Management Australia Ltd	Buy	2 808	900	Ordinary
01-10-14	ILIRS Wealth Management Australia Ltd	Cell	4 487	(464)	Ordinan
01-Jul-14	UBS Securities Australia Ltd	Buv	9826	1 013	Ordinary
01-101-14	UBS Securities Australia I td	Sell	1 043	(108)	Ordinary
01-14-14	ILIRS Securities Australia Ltd	2 8	OUE.	32	Ordinary
01-101-14	LIBS Securities Australia I td	9.15	126	100	Ordinary
10,17	HIDS Convision Australia Lad		27.70	2,20	Ordinally Ordinally
01-1-1-1-1	LIBS Consistion Australia Ltd	And a	1,695	(32	Ordinary
11 17	I IDS Convision Australia Ltd	sos a	4,455	1000	Ordinary
11.10	Libe committee Australia Liu	ina s	93,134	00000	Orginary
01-101-14	UBS Securities Australia Ltd	any	000005	9/1/6	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Sell	32,496	(3,364)	Ordinary
71-JUI-14	UBS Securities Australia Ltd	Buy	47,720	4,940	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	33,907	3,510	Ordinary
01-Juj-14	UBS Securities Australia Ltd	Виу	89	7	Ordinary
01-Jui-14	UBS Securities Australia Ltd	Buy	1,448	150	Ordinary
01-jul-14	UBS Securities Australia Ltd	Buy	10	1	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
01-jul-14	UBS Securities Australia Ltd	Buy	10	ı	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	85	9	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	39	4	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	89	7	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	193	20	Ordinary
31-Jul-14	UBS Securities Australia Ltd	Buy	39	4	Ordinary
31-Jul-14	UBS Securities Australia Ltd	Buy	1,042	108	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	772	80	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	059'6	1,000	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	14,475	1,500	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	599	31	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	2,789	289	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	1,148	119	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Виу	23,411	2,426	Ordinary
01-301-14	UBS Securities Australia Ltd	Buy	3,850	400	Ordinary
1 11 14	LIDS Securities Australia Ltd	Áng	460'C	080	Ordinary
11-10-14	LIBC Counties Australia Ltd	and a	00,00	1 000	Ordinary
1 1 1 1	LIDE Commission Australia Ltd	in a	10,130	000,	Ordinary
01-304-14	LIDS Securities Australia Ltd	ybd.	/10,01	1,038	Ordinary
1 1 1 1	Tipe Contribution Australia Ltd	euy G	10,400	0 000	Crainary
01-301-14	Ups Securities Australia Ltd	anà	19,493	020,2	Ordinary
71-JUI-14	UBS Securities Australia Ltd	Buy	4,005	415	Ordinary
21-JUI-14	UBS Securities Australia Ltd	Buy	7/77	236	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	6,186	641	Ordinary
71-Jul-14	UBS Securities Australia Ltd	Buy	5,751	296	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	27,174	2,816	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	5,288	548	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	56,337	5,838	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	57,022	5,909	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buy	27,946	2,896	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	4,150	430	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	29,674	3,075	Ordinary
	UBS Securities Australia Ltd	Buy	4,584	475	Ordinary
	UBS Securities Australia Ltd	Buy	3,156	327	Ordinary
	UBS Securities Australia Ltd	Buy	28,342	2,937	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	5,317	551	Ordinary
01-Jui-14	UBS Securities Australia Ltd	Buy	3,156	327	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	12,564	1,302	Ordinary
01-303-14	UBS Securities Australia Ltd	Buy	3,156	327	Ordinary
01-juj-14	UBS Securities Australia Ltd	Buy	3,156	327	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buy	2,355	244	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	3,156	327	Ordinary
01-Jul-14	UBS Securities Australia Ltd	Buy	10	1	Ordinary
01-Jul-14	I IRC Congression Assettation 144	ă	107 0		
1	ספה הכתונוכה שתפוומות הנת	look	194'C	268	Ordinary

UBS Securities Australia Ltd	ing a	arc r	901	Ordinary
securities Australia Ltd	508	582	98	Ordinary
UBS Securities Australia Ltd	Buy	10,670	1,100	Ordinary
UBS Securities Australia Ltd	Buy	4,394	453	Ordinary
ecurities Australia Ltd	Buy	6,858	707	Ordinary
UBS Securities Australia Ltd	Buy	1,300	134	Ordinary
ecurities Australia Ltd	Buv	640	99	Ordinary
IRS Securities Australia I to	N.B.	191	2	Ordinary
IBC Con ritios Australia (10	Busy	5 025	213	racajoro
DOS SECUCIOS POSTIBILISTES	, no	20,02	2 2	Cocinaly
eculities Australia Ltd	, and	010,0	6/6	Cranary
Securities Australia Ltd	Buy	3,065	316	Ordinary
ecurities Australia Ltd	Buy	4,617	476	Ordinary
UBS Securities Australia Ltd	Buy	119	63	Ordinary
IBS Sequirities Australia Ltd	BIN	3 006	412	Ordinary
Securities Australia Etc	(ng	0000	21.5	(initial)
UBS Securities Australia Ltd	Buy	3,036	313	Ordinary
securities Australia Ltd	Buy	3,337	344	Ordinary
UBS Securities Australia Ltd	Buy	1,571	162	Ordinary
UBS Securities Australia Ltd	Buy	2,444	252	Ordinary
Securities Australia Ltd	Buv	4.608	475	Ordinary
Lat all and a section of the section	(S)	2000	2,70	Care in a
DDS Securifies Australia Ltd	han	2,330	/#7	Ordinary
UBS Securities Australia Ltd	Buy	931	96	Ordinary
UBS Securities Australia Ltd	Buy	9,370	996	Ordinary
IBS Societities Assertalia 14d	i di	705	683	Ordinary
מביסוונים שחפון פוופ דוכם	600	25,	3 3	in in in
UBS Securities Australia Ltd	Buy	4,918	/OS	Ordinary
Securities Australia Ltd	Buy	524	54	Ordinary
JBS Securities Australia Ltd	Buy	8,303	856	Ordinary
IBS Securities Australia (1d	Bir	1 057	100	Ordinary
100 C-1-11-1 A11-1	623	(50)	26,	
securities Australia Ltd	Buy	4,617	4/b	Ordinary
JBS Securities Australia Ltd	Buy	3,977	410	Ordinary
UBS Securities Australia Ltd	Buy	8,594	986	Ordinary
Securities Australia Ltd	Buy	1,862	192	Ordinary
IRS Securities Australia Ltd	Biv	708	73	Ordinary
DC Constitution Australia 14d	G G	8097	324	Ordina
Securities Australia Lite	600	000/1		Contrary of
UBS Securities Australia Ltd	Any	3,240	334	Ordinary
JBS Securities Australia Ltd	Buy	1,455	150	Ordinary
Securities Australia Ltd	Buy	3,977	410	Ordinary
IBS Sagurities Australia I to	216	43 650	A 500	Ordinan
Securities Australia Ltd	foa	200'5	2001	Ciging)
Opp Securities Australia Liu	and -	c l't	1,71	Colonialy
Securities Australia Ltd	Buy	10		Ordinary
UBS Securities Australia Ltd	Buy	243	25	Ordinary
UBS Securities Australia Ltd	SUV	4.821	497	Ordinary
Cocurition Australia 14d	Bush	2 257	enc.	Ordinany
Obs Securities Australia Litu	and .	7557	743	Guilary
Securities Australia Ltd	Buy	10	-	Ordinary
UBS Securities Australia Ltd	Buy	1,271	131	Ordinary
UBS Securities Australia Ltd	Buv	3.570	368	Ordinary
IBC Cocurition Australia 14d	Bing	8 170	858	Ordinan
Securities Australia Ltd	oo)	67170	3	Oldinal y
Obs securities Australia Lto	, and	C1C'1	00.	Ordinary
Securities Australia Ltd	Buy	1,/56	181	Ordinary
UBS Securities Australia Ltd	Buy	3,608	372	Ordinary
UBS Securities Australia Ltd	Buv	9.642	766	Ordinary
IBC Cocurities Assetzalia 14d	2010	502	13	Ordinan
The state of the s		200	6	The state of the s
טפט אראון פווש דיים	an)	27,74	7/6	Genaly
securities Australia Ltd	any	76	2	Ordinary
UBS Securities Australia Ltd	Buy	7,954	820	Ordinary
UBS Securities Australia Ltd	Buy	5,122	528	Ordinary
BS Securities Australia Ltd	Buy	6.295	649	Ordinary
Committee Assertable 144	, Q	0000	QV	Ordina
Coculting Australia 144	, a	21 004	000 7	Codings
Obs securities Australia ctu	had	#00'1 /	070'/	Collinary :
UBS Securities Australia Ltd	Buy	1,121,398	115,608	Ordinary
UBS Securities Australia Ltd	Buy	514	53	Ordinary
UBS Securities Australia Ltd	Buv	275.228	28.374	Ordinary
IDS Committee Australia 14d	6	57 230	008 3	Ordinan
Securities Australia Ltd	śna	077'/6	500,0	Clumary
UBS Securities Australia Ltd	Buy	61	7	Ordinary
Securities Australia Ltd	Stock borrowed	ΨN.	35,000	Ordinary
FIDS Contrition Australia 14d	las	Cro r		
securities Australia Ltd			111111	C
		8/0'/	(732)	Ordinary

01-101-14 UBS 01-101-14 UBS 01-101-14 UBS 01-101-14 UBS 01-101-14 UBS 01-101-14 UBS 01-101-14 UBS 01-101-14 UBS	IRS Securities Australia Itd	Buy	3,156	327	Ordinary
1111111		buy	001.0	,75	Closiary
111111	S Son urities Australia 14d	200	24.4	218	Ordinan
	UBS Securities Australia Ltd	Buy	2.181	226	Ordinary
1	UBS Securities Australia Ltd	Buy	3,156	327	Ordinary
1111	S Securities Australia Itd	Riv	3 156	327	Ordinary
	S Securities Australia Ltd	Buv	3.860	400	Ordinary
	UBS Securities Australia Ltd	Buy	46	10	Ordinary
Ĺ	35 Securities Australia Ltd	Buy	3,156	327	Ordinary
	S Securities Australia Ltd	Buy	3,156	327	Ordinary
۳	35 Securities Australia Ltd	Buy	23,816	2.468	Ordinary
ľ	35 Securities Australia Ltd	Buv	6.041	929	Ordinary
=	UBS Securities Australia Ltd	Buv	5.317	551	Ordinary
1	S Securities Australia I td	Buy	280	20	Ordinary
╬	S Securities Australia I to	Riv	000	7	Ordinary
1	S Securities Australia Itd	Birv	2 760	286	Ordinacy
1	S Securities Australia Itd	Birv	58 740	5.087	Ordinary
01-1-10	S Societies Australia 11d	Ben	3 271	230	Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	MC Countries Australia 14d	600	331.6	500	Ordinary
5 2	Societies Australia Ltd	900	120	775	Oldinary California
5 2	ODD Securities Australia Ltd	pan,	2 456	1.00	Ordinary
П	s Securities Australia Ltd	and	3,136	37/	Ordinary
-1	UBS Securities Australia Ltd	Buy	77,764	2,359	Ordinary
١	iS Securities Australia Ltd	Buy	790,007	2,695	Ordinary
02-Jul-14 UE	iS AG London Branch	Buy	1,915	197	Ordinary
ij	UBS AG London Branch	Buy	4,879	205	Ordinary
ij	UBS AG London Branch	Buy	1,268	130	Ordinary
3	UBS AG London Branch	Buy	205	21	Ordinary
Γ	JBS AG London Branch	Buy	926	95	Ordinary
02-Jul-14 UE	S AG London Branch	Buy	1.843	189	Ordinary
Τ	I BS AG London Branch	Rin	917	70	Ordinary
T	I Do A Calcon Broads	Sing.	300	3.2	Ordinary
11 17 00	IDC ACT CALLED DESCRIPTION	6.0	2510	910	Ordinal y
T	S AG LOIDON DISTILL	ana d	2,316	007	Oldinary
7	UBS AG London Branch	Buy	1,845	681	Ordinary
02-Jul-14 UE	UBS AG London Branch	Buy	156	16	Ordinary
	JBS AG London Branch	Buy	449	46	Ordinary
02-Jul-14 UE	JBS AG London Branch	Buy	420	43	Ordinary
Š	UBS AG London Branch	Buy	2,482	254	Ordinary
3	UBS AG London Branch	Buy	2,403	246	Ordinary
Ë	UBS AG London Branch	Buy	2776	233	Ordinary
	S AG London Branch	Buy	2 052	210	Ordinary
	I BC AG London Branch	Reny	1 133	116	Ordinan
1	IBS AG London Branch	, and	4 685	ORV	Ordinary
5 =	DO AC LONGO DOSAN	600	500,1	200	Ordinary Ordinary
5 :	S AG London Branch	ána.	1,0/3	02 5	Ordinary
Ĭ,	UBS AG London Branch	Buy	117	12	Ordinary
Ď	JBS AG London Branch	Buy	1,014	104	Ordinary
3	S AG London Branch	Buy	478	49	Ordinary
3	UBS AG London Branch	Buy	2,077	213	Ordinary
1	JBS AG London Branch	Buy	1.609	165	Ordinary
02-10-14 02-10-14	RS AG London Branch	Pi v	67	5	Ordinary
Ť	DO AC ECTION DISCH	600	64.	2 2	Ordinary Ordinary
1	S AG LOTIGOTI DIARIGI	anà.	77/	* ;	Courally Courally
02-Jul-14 UB	UBS AG London Branch	Buy	156	91	Ordinary
	S AG London Branch	Buy	322	33	Ordinary
	S AG London Branch	Buy	88	6	Ordinary
15	JBS AG London Branch	Boy	1.470	151	Ordinary
ءُ ا	BS AG London Branch	Buy	935	ď	Ordinary
T	4	100	000	2,1	ordinary.
an #1-Inf-70	CBS AG LOTIGOTI BLATICTI	Ana -	061,1	2	Ciginaly
٦	S AG London Branch	Buy	10/	7/	Ordinary
ÿ	JBS AG London Branch	Buy	1,314	135	Ordinary
<u>5</u>	UBS AG London Branch	Buy	992	102	Ordinary
	JBS AG London Branch	Buy	61	2	Ordinary
02-Jul-14 UB	JBS AG London Branch	Buy	107	11	Ordinary
f	JBS AG London Branch	Buv	87	6	Ordinary
Ť	IPS ACTURACY Branch	and a	123		Ordinan
3 5	o Ad London plantal	ana.	100	ñ	Ordinary
3	UBS AG London Branch	gnà	/8/	× :	Ordinary
5	JBS AG London Branch	Buy	970	100	Ordinary
ٍ≝	S AG London Branch	Buy	19	2	Ordinary
20.			1	1	65.50

1) In 14 I IDC AC Landon Branch	Buv	651	67	Ordinary
1		600	ò	
02-Jul-14 UBS AG London Branch	Buy	651	29	Ordinary
	Buy	379	39	Ordinary
	Buy	369	38	Ordinary
2-Jul-14 UBS AG London Branch	Buy	194	50	Ordinary
	Buv	116	12	Ordinary
UBS AG London Branch	Buv	1.445	149	Ordinary
UBS AG London Branch	Biv	883	6	Ordinary
UBS AG London Branch	Buv	2.364	244	Ordinary
UBS AG London Branch	Buv	853	88	Ordinary
02-Jul-14 LIBS AG London Branch	BIN	069	79	Ordinary
L	Bank	766	2 02	Ordinal y
T	626	20	n a	Ordinary
Ϊ	ng	/8	5	Orginary
Ť	Buy	67	5	Ordinary
UBS AG London Branch	Buy	1,260	130	Ordinary
12-Jul-14 UBS AG London Branch	Buy	475	49	Ordinary
	Buy	252	56	Ordinary
Ē	Buv	87	6	Ordinary
UBS AG London Branch	Biv	48	u	Ordinary
UBS AG London Branch	Bity	002	, 5	Ordinary
L		200	,	dia di
Opp Ad Lolldon Branch	Áng	05/	0 (Ordinary
UZ-JUI-14 UBS AG LONGON Branch	Buy	61	2	Ordinary
1	Buy	29	m	Ordinary
UBS AG London Branch	Buy	3,266	337	Ordinary
	Buy	2,141	221	Ordinary
	Buy	48	2	Ordinary
22-Jul-14 UBS AG London Branch	Buy	455	47	Ordinary
ı	Buv	455	47	Ordinary
ı	BitA	89		Croinso
Ť	6.0	1 570	, 25	Coming
The section of the se	S C C	0/6,1	102	Quillay.
UBS AG LORGON Branch	Buy	300	5	Ordinary
UBS AG London Branch	Buy	126	13	Ordinary
UBS AG London Branch	Buy	1,570	162	Ordinary
UBS AG London Branch	Buy	2,791	288	Ordinary
UBS AG London Branch	Buy	223	23	Ordinary
UBS AG London Branch	Buy	00€	31	Ordinary
UBS AG London Branch	Buy	1,502	155	Ordinary
UBS AG London Branch	Buy	8.576	882	Ordinary
UBS AG London Branch	Buv	1,502	155	Ordinary
UBS AG London Branch	Bary	136	14	Ordinary
UBS AG London Branch	Buv	649	67	Ordinary
I IRS AG London Branch	Brix	340	ýs YE	Ordinary
I IRS AG London Branch	Para Research	500	200	Ordinary
I like A C London Branch	ing a	2001	300	Ciclinary
UBS AG LONGON Branch	Buy	086,1	507	Ordinary
UBS AG London Branch	Buy	543	26	Ordinary
UBS AG London Branch	Buy	233	24	Ordinary
UBS AG London Branch	Buy	78	80	Ordinary
UBS AG London Branch	Buy	320	33	Ordinary
UBS AG London Branch	Buy	1,28,5	397	Ordinary
UBS AG London Branch	Buy	78	80	Ordinary
١	Buy	136	14	Ordinary
02-Jul-14 UBS AG London Branch	Buy	194	20	Ordinary
	Buy	330	34	Ordinary
UBS AG London Branch	Buv	834	98	Ordinary
02-Jul-14 UBS AG London Branch	Buv	50.101	5.165	Ordinary
Ť	81%	10.495	1.082	Ordinary
Ī	818	3 372	348	Ordinary
UBS AG London Branch	Bliv	378	200	Ordinan
I IBC AG London Branch	(C)	0.00	100	Grainary
Line AG London Branch	Sno	616,1	06.	Ordings
02-Mil 14 LIBS AG LONGON Branch	Sing Prince	612	77	Ordinary
OBS AG LOI don Brailich	, and	650	0 5	Orginary
UBS AG London Branch	Buy	3,643	376	Ordinary
UBS AG London Branch	Виу	3,779	330	Ordinary
UBS AG London Branch	Buy	161	2	Ordinary
UBS AG London Branch	Buv	8.538	882	Ordinary

Buy 378 Buy 1,076 Buy 1,076 Buy 1,076 Buy 2,221 Buy 2,221 Buy 2,304 Buy 1,174 Buy 1,174 Buy 1,174 Buy 1,174 Buy 1,174 Buy 1,175 Buy 1,174 Buy 1,175 Buy 1,175 Buy 1,175 Buy 1,175 Buy 1,175 Buy 1,175 Buy 2,266 Buy 2,666 Buy 2,787 Buy	<u></u>	T	Γ.	Γ.	Τ.	T	T	Ī	Τ.	I.	Γ.	Τ.		Ī.		Γ			<u>.</u>	T	Τ	T	Ī.			T	Τ	T	Γ						Γ	T.		7	Т	Ι			\top		Τ	Τ			П	Т	7		1	Τ		ľ	Ι	П	I
Buy 378 Buy 1,076 Buy 1,076 Buy 1,06 Buy 2,250 Buy 2,250 Buy 2,304 Buy 1,174 Buy 1,174 Buy 1,176 Buy 2,666 Buy 1,176 Buy 2,666 Buy 2,787 Buy 1,176 Buy 1,176 Buy 2,666 Buy 2,666 Buy 2,787 Buy 1,136 Buy 1,136 Buy	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordinan	Ordina	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	, and in a									
Buy Buy	39	111	129	229	91	114	121	247	14	32	20	29	21	23	18	18	95	47	75	/8/	15	28	45	11	9	17	12	2 06	82	29	15	137	213	855	3 =	287	254	254	384	576	57	143	28	117	4-	33	264	118	105	105	23	49	48	4 0	145	65	276	129	272
	378	1.076	1.250	2,221	883	1106	1174	5 306	136	310	194	281	204	223	175	175	543	456	728	103	155	563	437	107	58	165	117	874	962	282	146	1,330	2,068	8,302	107	2,787	2,466	2,466	3,729	5,593	223	1,389	272	1,136	1,10/	320	2,563	1,146	1,020	1,020	223	476	466	175	1.408	631	2,680	1,253	***
london Branch Lo	Buv	Buy	Buv	Buy	Buv	Rin	Rin	Briv	Buy	Buv	Buy	Buv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buv	Buy	Buy	Buy	Buy	Buy	Buy	Biv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Виу	Buy	Buy	Buy	8uy	Buy	Buy	Buy	Buy	Buy	Buy	-
	London Branch	London Branch	London Branch	UBS AG London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	UBS AG London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	UBS AG London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	London Branch	JBS AG London Branch	UBS AG London Branch	London Branch	London Branch	London Branch	UBS AG London Branch	London Branch	London Branch	London Branch	London Branch	UBS AG London Branch	UBS AG London Branch	JBS AG tondon Branch	London Branch	UBS AG London Branch	JBS AG London Branch							

	OBS AG LUIGUII DI AIICII				-
	URS AG London Branch	Sug.	1 557	107	Ordinary
Ш	UBS AG tondon Branch	Buy	184	5	Ordinary
02-Jul-14	UBS AG London Branch	Buy	77	00	Ordinary
I	UBS AG London Branch	Buv	203	212	Ordinary
4	UBS AG London Branch	Buy	716	74	Ordinary
П	UBS AG London Branch	Buy	36	4	Ordinary
	UBS AG London Branch	Buy	193	20	Ordinary
	UBS AG London Branch	Buy	116	12	Ordinary
ı	UBS AG London Branch	Buy	300	31	Ordinary
02-Jul-14	UBS AG London Branch	Buy	184	19	Ordinary
1	UBS AG London Branch	Buy	19	2	Ordinary
1	UBS AG London Branch	Buy	184	19	Ordinary
02-Jul-14	UBS AG tondon Branch	Buy	523	54	Ordinary
-	UBS AG London Branch	Buy	1,056	109	Ordinary
ı	UBS AG London Branch	Buy	22,219	2,293	Ordinary
02-Jul-14	UBS AG London Branch	Buy	1,163	120	Ordinary
- 1	UBS AG London Branch	Buy	388	40	Ordinary
	UBS AG London Branch	Buy	194	20	Ordinary
	UBS AG London Branch	Buy	5,465	264	Ordinary
	UBS AG London Branch	Buy	465	48	Ordinary
	UBS AG London Branch	Buy	59	3	Ordinary
12-Jul-14	UBS AG London Branch	Buy	1,287	133	Ordinary
	UBS AG London Branch	Buy	1,471	152	Ordinary
02-Jul-14	UBS AG London Branch	Buy	2,585	267	Ordinary
	UBS AG London Branch	Buy	136	14	Ordinary
	UBS AG London Branch	Buy	1,452	150	Ordinary
	JBS AG London Branch	Buy	155	16	Ordinary
2-Jul-14	JBS AG London Branch	Buy	252	56	Ordinary
ı	UBS AG London Branch	Buy	871	06	Ordinary
1	JBS AG London Branch	Buy	620	64	Ordinary
Г	JBS AG London Branch	Buv	26	101	Ordinary
Г	UBS AG London Branch	Buy	19	2	Ordinary
	UBS AG London Branch	Buv	397	41	Ordinary
02-Jul-14	UBS AG London Branch	Buy	1,491	154	Ordinary
	JBS AG London Branch	Buy	222	23	Ordinary
	JBS AG London Branch	Buy	2,127	220	Ordinary
	JBS AG London Branch	Buy	10,502	1,086	Ordinary
D2-Jul-14	JBS AG London Branch	Buy	1,122	116	Ordinary
02-Jul-14	UBS AG London Branch	Buy	1,874	194	Ordinary
Ī	UBS AG London Branch	Buy	309	32	Ordinary
Ī	JBS AG London Branch	Buv	1.835	190	Ordinary
02-Jul-14	JBS AG London Branch	Buy	821	28	Ordinary
Ť	IRS AG London Branch	100	831	50	Ordinally Ordinally
T	IRS AG London Branch	, and	100	000	Crainary
ı	IRS AG London Branch	, and	4,000	463	Ordinary
02-301-14	IBS AG LONDON Branch	and B. S.	4,2/0	447	Ordinary
ı	DES AG LONDON BRAILEN	png	611,7	/3/	Ordinary
02-301-14	UBS AG London Branch	Buy	688	76	Ordinary
	DE AG London Branch	aug B. E.	7/3	80	Ordinary
Ť	BS AC LONDON BIRTH	śng	060	7/2	Ordinary
Ť	USC AG London Branch	in a	0,240	173	Orginary
١	IRA AG London Branch	Sno on	200,	7/2	Orginary
ı	DE ACTION BISING	śna	17.	R S	Ordinary
02-301-14	IBS AG London Branch	, ind	1,471	152	Orginary
	LIBS AG London Branch	buy	702.51	202	Ordinary
ı	DS AG London Branch	, and	10,727	1,720	Cidinary
02-ful-14	URS AG London Branch	Bar.	040	0 0	Ordinary
T	BS AG London Branch	Bury	1 462	151	Ording
T	JRS AG London Branch	200	1 422	12.	Ordinary
T	IRS AG London Branch	, and	11 510	1100	Ordinary
Ť	UBS AG London Branch	S. S.	175.71	1 278	Ordinary
T	IRS AG London Branch	ed.	1,2,2,1	0/7/0	Ordinary
02-301-14	IBS AG London Branch	Sing Print	2,/10	087	Orginary
	OBS AC COLOCI BISING	any.	200	na i	Ordinary
0Z-JUJ-14	JBS AG London Branch	Buy	320	33	Ordinary

16 Ordinary		30 Ordinary	L	L	L	58 Ordinary				124 Ordinary		64 Ordinary			160 Ordinary	434 Ordinary		28 Ordinary		309 Ordinary		1,197 Ordinary	249 Ordinary	L		L	11 Ordinary		1,035 Ordinary	168 Ordinary			124 Ordinary			13 Ordinary		115 Ordinary		Ц	486 Ordinary	2 Ordinary	80 Ordinary					59 Ordinary		311 Ordinary			23 Ordinary	Ц	12 Ordinary
155	926	250	920	1,123	862	561	1881	781	581	1,200	348	620	755	136	1,549	107,4 OC	145	271	145	2,988	77	11,575	2,408	2,472	319	1,992	106	4,806	10,008	1,625	87	164	1,199	358	11,188	10501	155	1,111	1,584	1,159	4,695	310	774	4,158	13,306	4,158	1,344	571	505	2 040	155	348	222	174	116
Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Suy	Buy	Buv	Buy	Buy	Buy	Buy	Buy	auy Biz	Riv	Buy	Buy	Buy	Buy	Buy	Buv	Buy	Buy	Buy	Buy	Ruy	Blay	Buy	guy Bris	Buy	Riv	Buy	Buy	Buy	Buy											
UBS AG tondon Branch	UBS AG London Branch	USS AG London Branch	USS AG London Branch	UBS AG London Branch	UBS AG tondon Branch	UBS AG London Branch	UBS AG London Branch																																																
02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	-	02-Jui-14	- 1	1	1	02-301-14		02-Jul-14				02-Jul-14	1	1	02-Jul-14	02-Jul-14	Н		02-Jul-14	ł	02-Jul-14	02-Jui-14	02-Jul-14	02-Jul-14	02-Jul-14	02-Jul-14	02-101-14	02-Jul-14	02-Jul-14	02-Jul-14	 02-Jul-14														

185 Securities Australia Ltd 548 126	02-hil-14	(IRS Securities Australia Ital	lle Sell	1,057	(400)	ording,
LBS Securities Australia Ltd Sell 7.85 C.	4-	UBS Securities Australia Ltd	les.	3,000	(400)	Ordinary
LBS Securities Australia Ltd Sell 26.452 CQ LBS Securities Australia Ltd Sell 2373 CQ LBS Securities Australia Ltd Sell 7.391 CQ LBS Securities Australia Ltd Sell 2.00 CQ LBS Securities Australia Ltd Sell 3.00 CQ LBS Securities Australia Ltd Sell 1.00 CQ LBS Securities Australia Ltd Sell 1.00 CQ LBS Securities Australia Ltd Sell 1.05 CQ LBS Securities Australia Ltd Buy 2.00 CA LBS Securities Australia Ltd Buy 2.00	-14	UBS Securities Australia Ltd	Sell	982	(81)	Ordinary
USS Securities Australia Ltd Self 19,000 CE USS Securities Australia Ltd Self 19,000 CE USS Securities Australia Ltd Self 19,000 CE USS Securities Australia Ltd Self 204 204 USS Securi	02-Jul-14	UBS Securities Australia Ltd	Sell	26,452	(727.2)	Ordinary
UBS Securities Australia Ltd Sell 19,500 C C UBS Securities Australia Ltd Sell 5,539 Sell 5,539 UBS Securities Australia Ltd Sell 2,530 Sell 2,530 Securities Australia Ltd Sell 2,541 2,541 Sell 3,880 Securities Australia Ltd Sell 3,880 Securities Australia Ltd Sell 3,880 Securities Australia Ltd Sell 3,880 Securities Australia Ltd Sell 3,880 Securities Australia Ltd Sell 3,880 Securities Australia Ltd Sell 3,880 Securities Australia Ltd Sell 3,980 Securities Australia Ltd Sell Sell 3,980 Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Securities Australia Ltd Sell Sell Securities Australia Ltd Sell Securi	-14	UBS Securities Australia Ltd	Sell	213	(22)	Ordinary
UBS Securities Australia Ltd Sell 7.391 UBS Securities Australia Ltd Sell 3.78 UBS Securities Australia Ltd Sell 3.78 UBS Securities Australia Ltd Sell 3.78 UBS Securities Australia Ltd Sell 3.89 UBS Securities Australia Ltd Sell 3.78 UBS Securities Australia Ltd Sell 1.05 UBS Securities Australia Ltd Sell 1.05 UBS Securities Australia Ltd Buy 2.04 UBS Securities Australia Ltd Buy 2.93 1.05 UBS Securities Australia Ltd Buy 2.93 2.93 1.05	-14	UBS Securities Australia Ltd	Sell	19,400	(2.000)	Ordinary
USS Securities Australia Ltd Sell 5,599 USS Securities Australia Ltd Sell 378 USS Securities Australia Ltd Sell 378 USS Securities Australia Ltd Sell 378 USS Securities Australia Ltd Sell 380 USS Securities Australia Ltd Sell 1,057 USS Securities Australia Ltd Sell 1,057 USS Securities Australia Ltd Sell 1,057 USS Securities Australia Ltd Buy Sell 1,057 USS Securities Australia Ltd Buy Sell 1,367 373 USS Securities Australia Ltd Buy 1,367 373 173 USS Securities Australia Ltd Buy 1,367 373 <t< td=""><td>-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>7.391</td><td>(762)</td><td>Ordinary</td></t<>	-14	UBS Securities Australia Ltd	Sell	7.391	(762)	Ordinary
UBS Securities Australia Ltd Sell 378 UBS Securities Australia Ltd Sell 378 UBS Securities Australia Ltd Sell 3890 UBS Securities Australia Ltd Sell 407 UBS Securities Australia Ltd Buy 78-31 UBS Securities Australia Ltd Buy 75-37 UBS Securities Australia Ltd Buy 75-37 UBS Securities Australia Ltd Buy 75-37 UBS Securities Australia Ltd Buy	-14	UBS Securities Australia Ltd	روا	5 530	(571)	Ordinary
UBS Securities Autorials Ltd Sell 270 UBS Securities Autorials Ltd Sell 27 UBS Securities Autorials Ltd Sell 380 UBS Securities Autorials Ltd Sell 1,057 UBS Securities Autorials Ltd Sell 1,057 UBS Securities Autorials Ltd Sell 1,057 UBS Securities Autorials Ltd Since tectured NW UBS Securities Autorials Ltd Buy 7,543 UBS Securities Autorials Ltd Buy 7,543 UBS Securities Autorials Ltd Buy 7,573 UBS Securities Autorials Ltd Buy	14	PRS Securities Australia 11d	100	0000	(00)	Oldinary
UBS Securities Australia Ltd Sell 9.07 UBS Securities Australia Ltd Sell 3.880 UBS Securities Australia Ltd Sell 4.07 UBS Securities Australia Ltd Sell 4.07 UBS Securities Australia Ltd Sell 1,057 UBS Securities Australia Ltd Sell 2,044 UBS Securities Australia Ltd Buy 2,647 UBS Securities Australia Ltd Buy 2,624,17 UBS Securities Australia Ltd Buy 1,25,70 UBS Securities Australia Ltd	5	1 BC Cocurities Australia 14d	100	9/6	(60)	Cidicialy
UBS Securities Australia Ltd Sell 3.89 UBS Securities Australia Ltd Sell 1,055 UBS Securities Australia Ltd Sell 1,056 UBS Securities Australia Ltd Sell 1,056 UBS Securities Australia Ltd Sell 1,056 UBS Securities Australia Ltd Buy 2,044 UBS Securities Australia Ltd Buy 2,047 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy		1 100 Committee A retaining Ltd	100	107	(17)	Ordinary
UBS Securities Australia Ltd Sell 4.057 UBS Securities Australia Ltd Sell 1,057 UBS Securities Australia Ltd Sell 1,057 UBS Securities Australia Ltd Sinck borrowed NA UBS Securities Australia Ltd Sinck borrowed NA UBS Securities Australia Ltd Buy 7,843 UBS Securities Australia Ltd Buy 7,843 UBS Securities Australia Ltd Buy 2,967 UBS Securities Australia Ltd Buy 2,977 UBS Securities Australia Ltd	1	DESCRIPTION AUSTRALIA	Sell	6	(01)	Ordinary
UBS Securities Australia Ltd Sell 1,057 UBS Securities Australia Ltd Sell 1,057 UBS Securities Australia Ltd Sinck borrowed NAA UBS Securities Australia Ltd Buy 7,843 UBS Securities Australia Ltd Buy 29,647 UBS Securities Australia Ltd Buy 29,647 UBS Securities Australia Ltd Buy 29,637 UBS Securities Australia Ltd Buy 29,637 UBS Securities Australia Ltd Buy 29,507 UBS Securities Australia Ltd Buy 29,507 UBS Securities Australia Ltd Buy 23,270 UBS Securities Australia Ltd Buy 29,507 UBS Securities Australia Ltd Buy 23,270 UBS Securities Australia Ltd Buy 23,270 UBS Securities Australia Ltd Buy 23,270 UBS Securities Australia Ltd		UBS Securities Australia Ltd	Sell	3,880	(400)	Ordinary
UBS Securities Australia Ltd Sell 1,057 UBS Securities Australia Ltd Sell 1,058 UBS Securities Australia Ltd Sing 1,056 UBS Securities Australia Ltd Buy 204 UBS Securities Australia Ltd Buy 7,843 UBS Securities Australia Ltd Buy 7,873 UBS Securities Australia Ltd Buy	-14	UBS Securities Australia Ltd	Sell	407	(42)	Ordinary
UBS Securities Australia Ltd Sell 1,106 UBS Securities Australia Ltd Sinch Enrumed 1,106 UBS Securities Australia Ltd Shot 1,06 UBS Securities Australia Ltd Buy 7,843 UBS Securities Australia Ltd Buy 29,0417 UBS Securities Australia Ltd Buy 5,90 UBS Securities Australia Ltd Buy 3,92 UBS Securities Australia Ltd Buy 3,20 UBS Securities Australia Ltd Buy 3,20 UBS Securities Australia Ltd Buy 4,24,15 UBS Securities Australia Ltd Buy 4,24,15 UBS Securities Australia Ltd Buy 4,24,15 UBS Securities Australia Ltd Buy 4,22,17 UBS Securities Australia Ltd Buy 5,27 UBS Securities Australia Ltd Buy 7,5,73 UBS Securities Australia Ltd <t< td=""><td>-14</td><td>UBS Securities Australia Ltd</td><td>Sel!</td><td>1,057</td><td>(109)</td><td>Ordinary</td></t<>	-14	UBS Securities Australia Ltd	Sel!	1,057	(109)	Ordinary
UBS Securities Australia Ltd Stock borrowed NAR UBS Securities Australia Ltd Shock returned INAR UBS Securities Australia Ltd Buy 78,431 UBS Securities Australia Ltd Buy 78,431 UBS Securities Australia Ltd Buy 78,432 UBS Securities Australia Ltd Buy 78,432 UBS Securities Australia Ltd Buy 78,432 UBS Securities Australia Ltd Buy 81,6785 UBS Securities Australia Ltd Buy 81,6785 UBS Securities Australia Ltd Buy 83,152 UBS Securities Australia Ltd Buy 83,152 UBS Securities Australia Ltd Buy 428,413 UBS Securities Australia Ltd Buy 428,413 UBS Securities Australia Ltd Buy 428,413 UBS Securities Australia Ltd Buy 1,23,731 UBS Securities Australia Ltd Buy 43,200 UBS Securities Australia Ltd Buy 1,23,731 UBS Securities Australia Ltd Buy 1,23,731 UBS Securities	-14	UBS Securities Australia Ltd	Sel	1,106	(114)	Ordinary
USS Securities Australia Ltd Stock betrowed NAZ USS Securities Australia Ltd Buy 15,813 USS Securities Australia Ltd Buy 15,814 USS Securities Australia Ltd Buy 15,814 USS Securities Australia Ltd Sell 188,489 USS Securities Australia Ltd Buy 48,350 USS Securities Australia Ltd Buy 48,350 USS Securities Australia Ltd Buy 316,752 USS Securities Australia Ltd Sell 38,152 USS Securities Australia Ltd Sell 38,170 USS Securities Australia Ltd Buy 42,84,15 USS Securities Australia Ltd Buy 42,84,15 USS Securities Australia Ltd Buy 7,573 USS Securities Australia Ltd Buy 7,53,270 USS Securities Au	-14	UBS Securities Australia Ltd	Self	204	(21)	Ordinary
UBS Securities Australia Ltd Bloy Fig. 511 UBS Securities Australia Ltd Buy 7,843 UBS Securities Australia Ltd Buy 315,785 UBS Securities Australia Ltd Buy 315,785 UBS Securities Australia Ltd Buy 32,475 UBS Securities Australia Ltd Buy 4,28,415 UBS Securities Australia Ltd Buy 4,28,417 UBS Securities Australia Ltd Buy 4,29,71 UBS Securities Australia Ltd Buy 7,573 UBS Securities Australia Ltd Buy 1,29,71 UBS Securities Australia Ltd Buy 1,29,71 UBS Securities Australia Ltd Buy 1,29,73 UBS Securities Australia Ltd <td>-14</td> <td>UBS Securities Australia Ltd</td> <td>Stock borrowed</td> <td>ΝA</td> <td>29,193</td> <td>Ordinary</td>	-14	UBS Securities Australia Ltd	Stock borrowed	ΝA	29,193	Ordinary
UBS Securities Australia Ltd Buy 15,511 UBS Securities Australia Ltd Buy 7,843 UBS Securities Australia Ltd Buy 79,452 UBS Securities Australia Ltd Buy 29,452 UBS Securities Australia Ltd Buy 48,350 UBS Securities Australia Ltd Buy 48,350 UBS Securities Australia Ltd Buy 48,350 UBS Securities Australia Ltd Buy 36,175 UBS Securities Australia Ltd Buy 363,167 UBS Securities Australia Ltd Buy 48,8415 UBS Securities Australia Ltd Buy 48,8415 UBS Securities Australia Ltd Buy 12,873 UBS Securities Australia Ltd Buy 12,873 UBS Securities Australia Ltd Buy 19,370 UBS Securities Australia Ltd Buy 5,814 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd	-14	UBS Securities Australia Ltd	Stock returned	ΥN	(32,000)	Ordinary
UBS Securities Australia Ltd Buy 7,843 UBS Securities Australia Ltd Sell 188,489 UBS Securities Australia Ltd Buy 29,425 UBS Securities Australia Ltd Buy 48,580 UBS Securities Australia Ltd Buy 48,535 UBS Securities Australia Ltd Sell 2,437 UBS Securities Australia Ltd Sell 2,677 UBS Securities Australia Ltd Buy 48,515 UBS Securities Australia Ltd Sell 2,677 UBS Securities Australia Ltd Buy 2,673 UBS Securities Australia Ltd Buy 2,673 UBS Securities Australia Ltd Buy 12,873 UBS Securities Australia Ltd Buy 12,873 UBS Securities Australia Ltd Buy 12,973 UBS Securities Australia Ltd Buy 1,29,73 UBS Securities Australia Ltd Buy 1,29,73 UBS Securities Australia Ltd Buy 49,297 UBS Securities Australia Ltd Buy 1,37 UBS Securities Australia Ltd <t< td=""><td>-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>15.511</td><td>1,600</td><td>Ordinary</td></t<>	-14	UBS Securities Australia Ltd	Buy	15.511	1,600	Ordinary
UBS Securities Australia Ltd Buy 590,417 UBS Securities Australia Ltd Self 590,422 UBS Securities Australia Ltd Buy 16,785 UBS Securities Australia Ltd Buy 316,785 UBS Securities Australia Ltd Buy 363,162 UBS Securities Australia Ltd Self 2,907 UBS Securities Australia Ltd Buy 363,162 UBS Securities Australia Ltd Buy 363,162 UBS Securities Australia Ltd Buy 1,833,20 UBS Securities Australia Ltd Buy 1,833,20 UBS Securities Australia Ltd Buy 1,833,20 UBS Securities Australia Ltd Buy 1,833,70 UBS Securities Australia Ltd Buy 1,937,70 UBS Securities Australia Ltd Buy 1,937,70 UBS Securities Australia Ltd Buy 1,937,70 UBS Securities Australia Ltd Buy 1,317,70 UBS Securities Australia Ltd Buy 1,317,70 UBS Securities Australia Ltd Buy 1,327,70 UBS Securities A	1-14	UBS Securities Australia Ltd	Buv	7.843	808	Ordinary
UBS Securities Australia Ltd Sell 188,489 UBS Securities Australia Ltd Buy 316,785 UBS Securities Australia Ltd Buy 316,785 UBS Securities Australia Ltd Sell 24,335 UBS Securities Australia Ltd Buy 48,415 UBS Securities Australia Ltd Sell 2,207 UBS Securities Australia Ltd Sell 75,373 UBS Securities Australia Ltd Buy 33,270 UBS Securities Australia Ltd Buy 75,373 UBS Securities Australia Ltd Buy 12,97,311 UBS Securities Australia Ltd Buy 1,29,731 UBS Securities Australia Ltd Buy 1,29,731 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,930 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd	-14	UBS Securities Australia Ltd	Buv	590.417	60 991	Ordinary
UBS Securities Australia Ltd Sell 188,488 UBS Securities Australia Ltd Buy 316,785 UBS Securities Australia Ltd Sell 24,335 UBS Securities Australia Ltd Buy 48,350 UBS Securities Australia Ltd Buy 48,350 UBS Securities Australia Ltd Buy 24,337 UBS Securities Australia Ltd Buy 428,413 UBS Securities Australia Ltd Buy 428,413 UBS Securities Australia Ltd Buy 428,413 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,934 UBS Securities Australia Ltd Buy 5,937 UBS Securities Australia Ltd Buy 5,072 UBS Securities Australia Ltd Buy 5,072 UBS Securities Australia Ltd Buy 5,072 UBS Securities Australia Ltd Buy 1,43,136 UBS Securities Australia Ltd Buy 1,43,136 UBS Securities Australia Ltd	14	LIBS Securities Australia I td) le	20.452	(000 C)	Ordinary
UBS Securities Australia Ltd Bay 16,788 UBS Securities Australia Ltd Buy 48,350 UBS Securities Australia Ltd Sell 24,325 UBS Securities Australia Ltd Sell 2,907 UBS Securities Australia Ltd Buy 4,84,15 UBS Wealth Management Australia Ltd Buy 4,84,15 UBS Securities Australia Ltd Buy 1,29,170 UBS Securities Australia Ltd Buy 1,29,170 UBS Securities Australia Ltd Buy 1,29,170 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 4,30,500 UBS Securities Australia Ltd Buy 1,47,209 UBS Securities Australi	1.14	I IRS Sequities Australia 14d	100	204,021	(00000)	Cidilary
UBS Securities Australia Ltd Buy 18,780 UBS Securities Australia Ltd Sell 24,325 UBS Securities Australia Ltd Buy 48,325 UBS Securities Australia Ltd Sell 2,907 UBS Securities Australia Ltd Buy 428,415 UBS Securities Australia Ltd Buy 123,270 UBS Securities Australia Ltd Buy 123,270 UBS Securities Australia Ltd Buy 123,730 UBS Securities Australia Ltd Buy 123,730 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,729 UBS Securities Australia Ltd Buy 6,189 UBS Securities Australia Ltd Buy 1,27,209 UBS Securities Australia Ltd Buy 1,31,310 UBS Securities Australia Ltd <td></td> <td>I IDC Countries Australia 14d</td> <td>100</td> <td>100,403</td> <td>(19,405)</td> <td>Ordinary</td>		I IDC Countries Australia 14d	100	100,403	(19,405)	Ordinary
UBS Securities Australia Ltd Bluy 843.350 UBS Securities Australia Ltd Bluy 383.162 UBS Securities Australia Ltd Sell 2,907 UBS Securities Australia Ltd Bluy 478.415 UBS Securities Australia Ltd Bluy 478.415 UBS Securities Australia Ltd Bluy 478.417 UBS Securities Australia Ltd Bluy 478.417 UBS Securities Australia Ltd Bluy 129.73 UBS Securities Australia Ltd Bluy 129.170 UBS Securities Australia Ltd Bluy 129.73 UBS Securities Australia Ltd Bluy 5,818 UBS Securities Australia Ltd Bluy 1,317 UBS Securities Australia Ltd Bluy 1,43,17 UBS Securities Australia Ltd Bluy 1,42,17 UBS Securities Australi		LIDE Completes Australia Ltd	buy	316,785	32,559	Ordinary
UBS Securities Australia Ltd Stell 363,162 UBS Securities Australia Ltd Sell 2,907 UBS Securities Australia Ltd Sell 2,907 UBS Securities Australia Ltd Sell 7,573 UBS Securities Australia Ltd Sell 7,573 UBS Wealth Management Australia Ltd Buy 12,873 UBS Wealth Management Australia Ltd Buy 12,873 UBS Securities Australia Ltd Buy 12,973 UBS Securities Australia Ltd Buy 19,170 UBS Securities Australia Ltd Buy 19,370 UBS Securities Australia Ltd Buy 5,880 UBS Securities Australia Ltd Buy 5,900 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 1,171 UBS Securities Australia Ltd Buy 1,371 UBS Securities Australia Ltd </td <td>: :</td> <td>Libs constitute Australia Ltd</td> <td>and and</td> <td>48,350</td> <td>5,000</td> <td>Ordinary</td>	: :	Libs constitute Australia Ltd	and and	48,350	5,000	Ordinary
UBS Securities Australia Ltd Buy 25,72 UBS Securities Australia Ltd Sell 2,907 UBS Securities Australia Ltd Ener Into Derivative (12,853) 2,907 UBS Securities Australia Ltd Buy 428,415 UBS Securities Australia Ltd Buy 12,823 UBS Securities Australia Ltd Buy 122,731 UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 19,370 UBS Securities Australia Ltd Buy 19,370 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,072 UBS Securities Australia Ltd Buy 6,188 UBS Securities Australia Ltd Buy 6,188 UBS Securities Australia Ltd Buy 7,209 UBS Securities Australia Ltd Buy 7,138 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 1,570 UBS Securities Austra		COO DECONTRIES AUSTRAIN LIQ	Sell	24,325	(2,500)	Ordinary
UBS Securities Australia Ltd Stell 2.907 UBS Securities Australia Ltd Sell 75.3,075 UBS Securities Australia Ltd Sell 75.3,075 UBS Wealth Management Australia Ltd Buy 75.3,270 UBS Wealth Management Australia Ltd Buy 12,853 UBS Wealth Management Australia Ltd Buy 12,937 UBS Securities Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 193,600 UBS Securities Australia Ltd Buy 193,600 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,907 UBS Securities Australia Ltd Buy 5,007 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 1,1171 UBS Securities Australia Ltd Buy 1,138 UBS Securities Australia Ltd Buy 1,138 UBS Securities Australia Ltd Buy 1,134 UBS Securities Aus	<u>.</u>	UBS Securities Australia Ltd	and	363,162	37,473	Ordinary
UBS Securities Australia Ltd Bluy 783 (075) UBS Securities Australia Ltd Sell 763 (075) UBS Securities Australia Ltd Bluy 75.73 UBS Wealth Management Australia Ltd Bluy 12.87.31 UBS Wealth Management Australia Ltd Bluy 12.87.731 UBS Wealth Management Australia Ltd Bluy 12.97.731 UBS Securities Australia Ltd Bluy 12.97.731 UBS Securities Australia Ltd Bluy 19.370 UBS Securities Australia Ltd Bluy 5.818 UBS Securities Australia Ltd Bluy 5.0007 UBS Securities Australia Ltd Bluy 5.0007 UBS Securities Australia Ltd Bluy 5.0007 UBS Securities Australia Ltd Bluy 7.1.138 UBS Securities Australia Ltd Bluy 7.5.729 UBS Securities Australia Ltd Bluy 7.5.729 UBS Securities Australia Ltd Bluy 7.1.138 UBS Securities Australia Ltd Bluy 7.1.138 UBS Securities Australia Ltd Bluy 7.1.138	4	UBS Securities Australia Ltd	Sell	2,907	(300)	Ordinary
UBS Securities Australia Ltd Sell 75,375 UBS Wealth Management Australia Ltd Buy 7,573 UBS Wealth Management Australia Ltd Buy 7,573 UBS Wealth Management Australia Ltd Buy 7,573 UBS Securities Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 193,600 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 49,997 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 1,3,136 UBS Securities Australia Ltd Buy 1,43,138 UBS Securities Australia Ltd Buy 1,43,136 UBS Securities Australia Ltd Buy 1,43,136 UBS Securities Australia Ltd Buy 1,43,136 UBS Securities Australia Ltd Buy 1,520 UBS Securities Au	-14	UBS Securities Australia Ltd	Buy	428,415	44,101	Ordinary
UBS Wealth Management Australia Ltd Enter into Derivative (IVA UBS Wealth Management Australia Ltd Buy 12,833 UBS Wealth Management Australia Ltd Sell 23,270 UBS Securities Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 99,594 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,910 UBS Securities Australia Ltd Buy 49,597 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 25,729 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 135,236 UBS Securities Australia Ltd Buy 13,328 UBS Securities Australia Ltd Buy 13,328 UBS Securities Australia Ltd Buy 13,424 UB	-14	UBS Securities Australia Ltd	Sell	763,075	(78,778)	Ordinary
UBS Wealth Management Australia Ltd Sell 7573 UBS Wealth Management Australia Ltd Buy 12,883 UBS Wealth Management Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 193,600 UBS Securities Australia Ltd Buy 193,000 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 49,997 UBS Securities Australia Ltd Buy 5,729 UBS Securities Australia Ltd Buy 5,729 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 7,138 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 17,138 UBS Securities Australia Ltd Buy 17,138 UBS Securities Australia Ltd Buy 13,201 UBS Securities Australia Ltd Buy 13,201 UBS Securities Australia Ltd Buy 13,404 UBS Securities Australi	-14	UBS Securities Australia Ltd	Enter Into Derivative (ΑN	5,342	Options
UBS Wealth Management Australia Ltd Buy 12,853 UBS Wealth Management Australia Ltd Buy 12,370 UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 49,997 UBS Securities Australia Ltd Buy 49,997 UBS Securities Australia Ltd Buy 5,729 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 25,729 UBS Securities Australia Ltd Buy 25,729 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 143,136 UBS Securities Australia Ltd Buy 143,136 UBS Securities Australia Ltd Buy 143,209 UBS Securities Australia Ltd Buy 143,200 UBS Securities Australia Ltd Buy 143,200 UBS Securities Australi	1-14	UBS Wealth Management Australia Ltd	Sell	7,573	(283)	Ordinary
UBS Wealth Management Australia Ltd Self 23,270 UBS Securities Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 193,600 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 49,927 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 1,171 UBS Securities Australia Ltd Buy 1,172 UBS Securities Australia Ltd Buy 1,173 UBS Securities Australia Ltd	1.14	UBS Wealth Management Australia Ltd	Buv	12,853	1 329	Ordinary
UBS Securities Australia Ltd Buy 129,731 UBS Securities Australia Ltd Buy 99,984 UBS Securities Australia Ltd Buy 99,984 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,006 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 6,138 UBS Securities Australia Ltd Buy 47,229 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 13,804 UBS Securities Australia Ltd <t< td=""><td>-14</td><td>UBS Wealth Management Australia Ltd</td><td>راوا</td><td>23 270</td><td>(2016)</td><td>Ordinany</td></t<>	-14	UBS Wealth Management Australia Ltd	راوا	23 270	(2016)	Ordinany
UBS Securities Australia Ltd Buy 99,994 UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 193,60 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,000 UBS Securities Australia Ltd Buy 49,937 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 25,729 UBS Securities Australia Ltd Buy 2,729 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 13,138 UBS Securities Australia Ltd Buy 145,209 UBS Securities Australia Ltd Buy 13,138 UBS Securities Australia Ltd Buy 145,209 UBS Securities Australia Ltd Buy 13,238 UBS Securities Australia Ltd Buy 13,238 UBS Securities Australia Ltd Buy 5,008 UBS Securities Australia Ltd Buy 5,008 UBS Securities Australia Ltd <td< td=""><td>-14</td><td>LIBS Securities Australia I to</td><td>Rin</td><td>126 731</td><td>12 402</td><td>Colling of</td></td<>	-14	LIBS Securities Australia I to	Rin	126 731	12 402	Colling of
UBS Securities Australia Ltd Buy 129,170 UBS Securities Australia Ltd Buy 193,600 UBS Securities Australia Ltd Buy 193,600 UBS Securities Australia Ltd Buy 5,618 UBS Securities Australia Ltd Buy 49,937 UBS Securities Australia Ltd Buy 49,937 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 13,138 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 1,138 UBS Securities Australia Ltd Buy 1,138 UBS Securities Australia Ltd Buy 1,360 UBS Securities Australia Ltd Buy 1,360 UBS Securities Australia Ltd Buy 1,38 UBS Securities Australia Ltd Buy 1,360 UBS Securities Australia Ltd	1-14	LIBS Securities Australia I td	Buy	700 00	00000	ordinary.
UBS Securities Australia Ltd Buy 159,170 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,937 UBS Securities Australia Ltd Buy 5,008 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 7,072 UBS Securities Australia Ltd Buy 1,43,136 UBS Securities Australia Ltd Buy 1,43,136 UBS Securities Australia Ltd Buy 1,43,209 UBS Securities Australia Ltd Buy 1,45,209 UBS Securities Australia Ltd Buy 1,45,209 UBS Securities Australia Ltd Buy 1,45,209 UBS Securities Australia Ltd Buy 1,34,804 UBS Securities Australia Ltd Buy 1,8,44 UBS Securities Australia Ltd Buy 5,608 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd		1 100 Coomition Australia 14d	buy.	466,66	00000	Granary
UBS Securities Australia Ltd Buy 193,600 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 5,818 UBS Securities Australia Ltd Buy 49,907 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 7,171 UBS Securities Australia Ltd Buy 7,209 UBS Securities Australia Ltd Buy 7,171 UBS Securities Australia Ltd Buy 1,171 UBS Securities Australia Ltd Buy 1,171 UBS Securities Australia Ltd Buy 1,173 UBS Securities Australia Ltd Buy 1,173 UBS Securities Australia Ltd Buy 8,73 UBS Securities Australia Ltd Buy 8,73 UBS Securities Australia Ltd Buy 8,73 UBS Securities Australia Ltd Buy 2,200 UBS Securities Australia Ltd Buy 2,201 UBS Securities Australia Ltd Buy 2,201 UBS Securities Australia Ltd Buy	<u>.</u>	ODS SECULTIES AUSTRAIR LIG	any	0/1,671	13,344	Ordinary
UBS Securities Australia Ltd Bluy 5,818 UBS Securities Australia Ltd Buy 19,370 UBS Securities Australia Ltd Buy 49,997 UBS Securities Australia Ltd Buy 30,008 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 11,173 UBS Securities Australia Ltd Buy 11,373 UBS Securities Australia Ltd Buy 11,373 UBS Securities Australia Ltd Buy 145,209 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 136,20 UBS Securities Australia Ltd Buy 136,20 UBS Securities Australia Ltd Buy 18,144 UBS Securities Australia Ltd Buy 134,604 UBS Securities Australia Ltd Buy 134,604 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd	4	UBS Securities Australia Ltd	Buy	193,600	20,000	Ordinary
UBS Securities Australia Ltd Bluy 19.370 UBS Securities Australia Ltd Buy 50.007 UBS Securities Australia Ltd Buy 30.908 UBS Securities Australia Ltd Buy 30.908 UBS Securities Australia Ltd Buy 25,729 UBS Securities Australia Ltd Buy 25,729 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 1,171 UBS Securities Australia Ltd Buy 1,38,26 UBS Securities Australia Ltd Buy 1,38,26 UBS Securities Australia Ltd Buy 1,38,26 UBS Securities Australia Ltd Buy 1,34,20 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 1,34,44 UBS Securities Australia Ltd Buy 2,22,012 UBS Securities Australia Ltd Buy 2,22,012 UBS Securities Australia Ltd	-14	UBS Securities Australia Ltd	Buy	5,818	601	Ordinary
UBS Securities Australia Ltd Buy 50,007 UBS Securities Australia Ltd Buy 49,927 UBS Securities Australia Ltd Buy 5,118 UBS Securities Australia Ltd Buy 72,729 UBS Securities Australia Ltd Buy 73,739 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 13,138 UBS Securities Australia Ltd Buy 13,138 UBS Securities Australia Ltd Buy 13,138 UBS Securities Australia Ltd Buy 13,200 UBS Securities Australia Ltd Buy 136,200 UBS Securities Australia Ltd Buy 136,200 UBS Securities Australia Ltd Buy 136,200 UBS Securities Australia Ltd Buy 134,804 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd	-14	UBS Securities Australia Ltd	Buy	19,370	2,001	Ordinary
UBS Securities Australia Ltd Buy 49.997 UBS Securities Australia Ltd Buy 6.118 UBS Securities Australia Ltd Buy 6.118 UBS Securities Australia Ltd Buy 47.209 UBS Securities Australia Ltd Buy 11.171 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 5.072 UBS Securities Australia Ltd Buy 9.670 UBS Securities Australia Ltd Buy 9.738 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 9.738 UBS Securities Australia Ltd Buy 134,824 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 222,012 UBS Securities Australia Ltd <td< td=""><td>-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>200'05</td><td>5,166</td><td>Ordinary</td></td<>	-14	UBS Securities Australia Ltd	Buy	200'05	5,166	Ordinary
UBS Securities Australia Ltd Buy 30,908 UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 15,000 UBS Securities Australia Ltd Buy 134,804 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 2,22,017 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Bu	-14	UBS Securities Australia Ltd	Buy	49,997	5,165	Ordinary
UBS Securities Australia Ltd Buy 6,118 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 15,200 UBS Securities Australia Ltd Buy 15,200 UBS Securities Australia Ltd Buy 136,236 UBS Securities Australia Ltd Buy 134,804 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 222,012 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd	-14	UBS Securities Australia Ltd	Bov	30 908	3.193	Ordinary
UBS Securities Australia Ltd Buy 25,729 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 1,43,138 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 1,43,200 UBS Securities Australia Ltd Buy 1,43,200 UBS Securities Australia Ltd Buy 1,34,804 UBS Securities Australia Ltd Buy 1,34,804 UBS Securities Australia Ltd Buy 1,34,804 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 1,83,744 UBS Securities Australia Ltd Buy 2,20,012 UBS Securities Australia Ltd Buy 2,22,012 UBS Securities Australia Ltd Buy 2,22,012 UBS Securities Australia Ltd Buy 2,22,012 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia	-14	UBS Securities Australia Ltd	Ruy	6118	632	Ordinan
UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 47,209 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 16,200 UBS Securities Australia Ltd Buy 136,236 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 28,988 UBS Securities Australia Ltd Buy 28,388 UBS Securities Australia Ltd Buy 26,838 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd	14	UBS Securities Australia 4th	Bury	75 720	3 650	Ordinary
UBS Securities Australia Ltd Buy 41,709 UBS Securities Australia Ltd Buy 11,171 UBS Securities Australia Ltd Buy 143,138 UBS Securities Australia Ltd Buy 5,072 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 157,138 UBS Securities Australia Ltd Buy 134,804 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 18,744 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 183,782 UBS Securities Australia Ltd Buy 222,012 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd		The second secon	on's	67/67	950'7	Ordinary
UBS Securities Australia Ltd Bby 14,71 UBS Securities Australia Ltd Buy 143,338 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 9,570 UBS Securities Australia Ltd Buy 71,138 UBS Securities Australia Ltd Buy 71,138 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 8,840 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 138,762 UBS Securities Australia Ltd Buy 220,101 UBS Securities Australia Ltd Buy 28,808 UBS Securities Australia Ltd Buy 28,808 UBS Securities Australia Ltd Buy 28,808 UBS Securities Australia Ltd Buy 28,818 UBS Securities Australia Ltd Buy 28,322,001 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd <	4	UBS Securities Australia Ltd	Buy	47,209	4,877	Ordinary
UBS Securities Australia Ltd Bby 143,138 UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 1,672 UBS Securities Australia Ltd Buy 1,738 UBS Securities Australia Ltd Buy 71,138 UBS Securities Australia Ltd Buy 134,804 UBS Securities Australia Ltd Buy 1,644 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 22,011 UBS Securities Australia Ltd Buy 266,388 UBS Securities Australia Ltd Buy 31,421 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd	4 :	UBS Securities Australia Ltd	Buy	11,171	1,154	Ordinary
UBS Securities Australia Ltd Buy 2,072 UBS Securities Australia Ltd Buy 9,670 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 17,138 UBS Securities Australia Ltd Buy 9,736 UBS Securities Australia Ltd Buy 9,736 UBS Securities Australia Ltd Buy 8,844 UBS Securities Australia Ltd Buy 5,806 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 222,002 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 222,002 UBS Securities Australia Ltd Buy 4,844 UBS Securities Australia Ltd Buy 4,68,388 UBS Securities Australia Ltd Buy 4,68,300 UBS Securities Australia Ltd	114	UBS Securities Australia Ltd	Buy	143,138	14,787	Ordinary
UBS Securities Australia Ltd Bby 9,670 UBS Securities Australia Ltd Buy 145,200 UBS Securities Australia Ltd Buy 71,138 UBS Securities Australia Ltd Buy 136,236 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 8,808 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840	-14	UBS Securities Australia Ltd	Buy	2,072	214	Ordinary
UBS Securities Australia Ltd Bby 145,200 UBS Securities Australia Ltd Buy 71,138 UBS Securities Australia Ltd Buy 136,236 UBS Securities Australia Ltd Buy 134,804 UBS Securities Australia Ltd Buy 1,948 UBS Securities Australia Ltd Buy 81,944 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 2,201 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 268,368 UBS Securities Australia Ltd Buy 268,368 UBS Securities Australia Ltd Buy 4,840	03-Jut-14	UBS Securities Australia Ltd	Buy	0/9'6	666	Ordinary
UBS Securities Australia Ltd Buy 71,138 UBS Securities Australia Ltd Buy 71,138 UBS Securities Australia Ltd Buy 9,736 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 5,806 UBS Securities Australia Ltd Buy 5,806 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 268,386 UBS Securities Australia Ltd Buy 4,840	-14	UBS Securities Australia Ltd	Buv	145.200	15.000	Ordinary
UBS Securities Australia Ltd Buy 136,236 UBS Securities Australia Ltd Buy 134,804 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 26,388 UBS Securities Australia Ltd Buy 26,388 UBS Securities Australia Ltd Buy 26,83,88 UBS Securities Australia Ltd Buy 4,840	-14	UBS Securities Australia Ltd	Buv	71.138	7 349	Ordinary
UBS Securities Australia Ltd Buy 134,820 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,849 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840	-14	UBS Securities Australia Ltd	Buy	136,236	AZO A1	Ordinary
UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 9,738 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840	3	I IDC Convision Australia Ind	6	200,000	† 00 c	- G
UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 2,008 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 31,421		TIDE Contribut Australia 144	oni,	134,804	976'51	Orginary
UBS Securities Australia Ltd Buy 81,844 UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 138,782 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840	# :	UBS Securities Australia Lto	Buy	9,738	1,006	Ordinary
UBS Securities Australia Ltd Buy 18,740 UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 31,421	-14	UBS Securities Australia Ltd	Buy	81,844	8,455	Ordinary
UBS Securities Australia Ltd Buy 5,808 UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 31,421	-14	UBS Securities Australia Ltd	Buy	18,740	1,936	Ordinary
UBS Securities Australia Ltd Buy 22,012 UBS Securities Australia Ltd Buy 138,782 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 31,421	-14	UBS Securities Australia Ltd	Buy	5.808	009	Ordinary
UBS Securities Australia Ltd Buy 138,782 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 222,001 UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 31,421	-14	IUBS Securities Australia Ltd	Riv	22.012	2.274	Ordinary
100 100	14	II IRS Societies Australia (1d	P	138 782	14 227	Ordinary
UBS Securities Australia Ltd Buy Z68,388 UBS Securities Australia Ltd Buy 7,88,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 31,421	: :	CDS Securities Australia Ltd	Buy	138,/82	14,337	Ordinary
UBS Securities Australia Ltd Buy 268,388 UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 31,421	4	UBS Securities Australia Ltd	Buy	222,001	22,934	Ordinary
UBS Securities Australia Ltd Buy 4,840 UBS Securities Australia Ltd Buy 31,421		UBS Securities Australia Ltd	Buy	268,388	27,726	Ordinary
UBS Securities Australia Ltd Buy 31,421		UBS Securities Australia Ltd	Buy	4,840	200	Ordinary
	l	UBS Securities Australia Ltd	Buy	31,421	3,246	Ordinary
I IBC Cocuration Australia 140	L	I IBC Cocurition Americal and	7.00	240,100	25000	0 0
UDD SECUTIVES AUSTRAIN LTG BUY 349,195	١	חסט ספרתווובא אחמוומוומ דוח	Buy	343,130	35,0/4	Ordinary
l nosio					733	Ordinan

02-Jul-14		-			
02-101-14	USS And Loridon Branch	Stock horroward	80,478	8,254	Ordinary
02-Jul-14	UBS Securities Australia Ltd	Buy	565.5		Ordinary
02-Jul-14	UBS Securities Australia Ltd	Sell	5,805		
02-lul-14	I IBS Securities Australia I to	2	79 391	ľ	
22-4-14	I IBS Societies Australia I to	وا	55,53		Ordinary
111	IDO Contribution Australia (4)	total	130.0		O Garany
11 107 20	COO Secretaria Acada and Coo	100	102,6		
1	Obs securities Australia Ltd	- Sei	510,11		
02-JUI-14	UBS Securities Australia Ltd	ie.	756'6/	(8,254)	
02-Jul-14	UBS Securities Australia Ltd	Buy	344,892	35,568	
- 1	UBS Securities Australia Ltd	Sell	9,726	(1,000)	
2-Jul-14	UBS Securities Australia Ltd	Sell	1,221	(126)	Ordinary
	UBS Securities Australia Ltd	Buy	94,310		
	UBS Securities Australia Ltd	Sell	114,417	(13,865)	Ordinary
02-Jul-14	UBS Securities Australia Ltd	Buy	406,490		
2-Jul-14	UBS Wealth Management Australia Ltd	Sell	2,282		
22-Jul-14	UBS Wealth Management Australia Ltd	Buv	1.867		l
1-14	I IRS Wealth Management Australia Ltd	رواز	2 127	(022)	Ordinan
02-1-1-14	IRS Wealth Management Australia Ltd	Ris	2 009		
1.14	I IRC Wealth Management Australia Ltd	Rin	009		Ordinary
2.101.17	TBC Washing Management Australia Ltd	600	200 6		Cyclinary
02-30F14	Obs. Wealth Mahagement Australia Ltd	kng.	0,990	0.4	Orginary
*	Obs Securities Australia Ltu	, and	24,234		Ordinary
JZ-JUF-14	UBS Securities Australia Ltd	Buy	1,725		Ordinary
-Jui-14	UBS Securities Australia Ltd	Sell	5,359		Ordinary
1-14	UBS Securities Australia Ltd	Sell	27,054		
02-Jul-14	UBS Securities Australia Ltd	Sell	52,210	(5,388)	Ordinary
22-Jul-14	UBS Securities Australia Ltd	Sell	38,043		Ordinary
02-Jul-14	UBS Securities Australia I td	و	33 721		
72-liil-14	PR Secritities A Istralia 12d	Sell	7 025		l
	TIDS Counties Australia 14d	130	700 \$		Collinary
	DISTRIBUTED CONTROL CON	11-0	(55,1	\$	dom and
-	Obsiderations Australia Ltd	Dell	050,00	9	Ordinary
72-Jul-14	UBS Securities Australia Ltd	lles	2,103		١
02-Jul-14	UBS Securities Australia Ltd	Sell	13,440	(1,387)	Ordinary
32-Jul-14	UBS Securities Australia Ltd	Sell	148,305		Ordinary
32-Jul-14	UBS Securities Australia Ltd	Sell	256,659		
10-14	UBS Securities Australia Ltd	Sell	9.186	(848)	Ordinary
-1111-14	I IRS Securities Australia I to	lla)	10		Ordinan
	The Commission Assessed and		E0+		diam's
11.17.77	Obsiderations Australia Etc) all	107	(42)	Ordinary
-Jui-14	UBS Securities Australia Ltd	Seil	398	(41)	Ordinary
02-Jul-14	UBS Securities Australia Ltd	Sell	68	(2)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	27,150	(2,799)	Ordinary
1-14	UBS Securities Australia Ltd	Sell	398	(11)	Ordinary
02.1.1.1.4	ADC Counties Australia Ltd	= 3	308	(44)	Cardioso
<u>.</u>	ODS Securities Adstraile Ltd	130	060	(14)	Ordinary
)2-Jul-14	UBS Securities Australia Ltd	Seli	107	(11)	Ordinary
12-Jul-14	UBS Securities Australia Ltd	Sell	960'1	(113)	Ordinary
12-14-14	LIRS Sequenties Australia 14d	Sell	398	(17)	Ordinary
	100 Commission A. (America)	200	2	()	(min)
02-301-14	UDS SECULITIES AUSTRAINA LEU	Sell	28	(a)	Ordinary
1-14	UBS Securities Australia Ltd	Sell	97	(01)	Ordinary
72-Jul-14	UBS Securities Australia Ltd	Sell	58	(9)	Ordinary
)2-Jul-14	UBS Securities Australia Ltd	38	065.05	(3.133)	Ordinary
02-hil-14	LIBS Societies Australia Ltd	= 9	1.096	(113)	dedical
	COO Securities Australia Lite	100	060,1	(01/)	Ciciliary
-	UBS Securities Australia Ltd	Sell	485	(nc)	Ordinary
	UBS Securities Australia Ltd	Sell	388	(40)	Ordinary
02-Jul-14	UBS Securities Australia Ltd	= 35	175	(18)	Ordinary
ı	UBS Securities Australia Ltd	Sell	398	((41)	Ordinary
	18S Securities Australia 11d	No.	10	(5)	Ordinan
1.1.4	LIDE Commission A resemble 1 and	0-1	000	î ĉ	diam'r.
*	Obsider Australia Ltd	E C	67	6	Ordinary
02-Jui-14	UBS Securities Australia Ltd	Seil	398	(41)	Ordinary
J-14	UBS Securities Australia Ltd	Sell	398	(41)	Ordinary
02-Jul-14	UBS Securities Australia Ltd	Sell	446	(46)	Ordinary
02-Jul-14	UBS Securities Australia Ltd	Sell	427	(44)	Ordinary
02-4ul-14	I IRC Countries Australia I tol	Sell	503	(55)	Ordinary
	ODS Securities Australia Ltd	1361	040	(00)	Ordinary
UZ-JUI-14	UBS Securities Australia Ltd	Sell	42,438	(4,3/5)	Ordinary
1-14	UBS Securities Australia Ltd	Sell	3,046	(314)	Crainson
11 11					Cicialy
	1 (B) Contribute Australia 140	170	5 245	(100)	Ordinary
02-JUL 14	UBS Securities Australia Ltd	Sell	5,345	(551)	Ordinary

URS Securities Automise a Lub Buy 16,572 1,712 URS Securities Automise a Lub Buy 1,657 1,712 URS Securities Automise a Lub Buy 2,458 2,573 URS Securities Automise a Lub Buy 2,439 2,573 URS Securities Automise a Lub Buy 2,439 7,573 URS Securities Automise a Lub Buy 2,439 7,571 URS Securities Automise a Lub Buy 2,589 7,77 URS Securities Automise a Lub Buy 2,589 7,87 URS Securities Automise a Lub Buy 2,589 2,64 URS Securities Automise a Lub Buy 3,622 2,64 URS Securities Automise a Lub Buy 3,622 2,64 URS Securities Automise a Lub Buy 3,622 3,64 URS Securities Automise a Lub	03-Jul-14	Obs Securities Australia Lid	500			
605 Securities Australia Lid Buy 6,000 10,000	11 14	UBS Securities Australia Ltd	gng	4,034	5/1	Ordinary
(MS) Securities Australia Ltd Buy 2,648 2,558 (MS) Securities Australia Ltd Buy 2,648 2,553 6,77 (MS) Securities Australia Ltd Buy 2,653 6,77 10,05 1,10 (MS) Securities Australia Ltd Buy 7,667 7,71 1,461 1,161 1,11 1,11 1,11 1,11 1,10	1111	UBS Secontiles Australia Ltd	Boy	7/5'01	1,712	Ordinary
USS Securities Australia Ltd Buy 1,065 105 1	1	ODS SECURITIES AUSTRALIA LIQ	buy	090'9	979	Ordinary
USS Securities Australia Ltd Buy 1,005 1	JUI-14	UBS Securities Australia Ltd	Buy	2,468	255	Ordinary
USS Securities Australia Lud Buy 6,553 2,51 USS Securities Australia Lud Buy 6,553 7,71 USS Securities Australia Lud Buy 1,641 1,418 USS Securities Australia Lud Buy 1,641 1,448 USS Securities Australia Lud Buy 1,641 1,448 USS Securities Australia Lud Buy 1,729 1,441 USS Securities Australia Lud Buy 1,729 1,441 USS Securities Australia Lud Buy 1,729 1,441 USS Securities Australia Lud Buy 1,729 1,442 USS Securities Australia Lud Buy 2,052 2,934 USS Securities Australia Lud Buy 3,072 3,936 USS Securities Australia Lud Buy 5,072 3,071 USS Securities Australia Lud <th< td=""><td>101-17</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1,065</td><td>110</td><td>Ordinary</td></th<>	101-17	UBS Securities Australia Ltd	Buy	1,065	110	Ordinary
UBS Securities Australia Lud Buy 1,553 677 UBS Securities Australia Lud Buy 7,667 791 UBS Securities Australia Lud Buy 1,466 1,513 UBS Securities Australia Lud Buy 1,614 1,461 UBS Securities Australia Lud Buy 2,503 273 UBS Securities Australia Lud Buy 3,872 400 UBS Securities Australia Lud Buy 3,872 40 UBS Securities Australia Lud Buy 3,872 40 UBS Securities Australia Lud Buy 2,826 2,54 UBS Securities Australia Lud Buy 1,925 1,13 UBS Securities Australia Lud Buy 1,243 1,45 UBS Securities Australia Lud Buy 1,243 1,45 UBS Securities Australia Lud Buy	-101	UBS Securities Australia Ltd	buy	2,439	757	Ordinary
UBS Securities Autorialia Ltd Buy 1,646 UBS Securities Autorialia Ltd Buy 1,646 UBS Securities Autorialia Ltd Buy 1,141 1,461 UBS Securities Autorialia Ltd Buy 5,024 5,20 UBS Securities Autorialia Ltd Buy 2,022 2,23 UBS Securities Autorialia Ltd Buy 2,022 2,23 UBS Securities Autorialia Ltd Buy 2,022 2,03 UBS Securities Autorialia Ltd Buy 8,02 2,03 UBS Securities Autorialia Ltd Buy 8,03 3,02 UBS Securities Autorialia Ltd Buy 3,03 3,03 UBS Securities Autorialia Ltd Buy 3,03 3,03 UBS Securities Autorialia Ltd Buy 3,03 3,03 UBS Securities Autorialia Ltd Buy 3,03 3,04 UBS Securities Autorialia Ltd Buy 3,04 3,05 UBS Securities Autorialia Ltd Buy 3,04 3,05 UBS Securities Autorialia Ltd Buy 3,04 3,05	41-ID	UBS Securities Australia Ltd	Buy	6,553	677	Ordinary
UBS Scorrites Australia Ltd Buy 1,148 1,148 UBS Scorrites Australia Ltd Buy 5,042 2,56 UBS Scorrites Australia Ltd Buy 5,042 2,56 UBS Scorrites Australia Ltd Buy 5,042 2,53 UBS Scorrites Australia Ltd Buy 5,075 2,13 UBS Scorrites Australia Ltd Buy 8,075 3,075 UBS Scorrites Australia Ltd Buy 8,075 3,075 UBS Scorrites Australia Ltd Buy 8,075 3,078 UBS Scorrites Australia Ltd Buy 8,075 3,078 UBS Scorrites Australia Ltd Buy 8,075 3,078 UBS Scorrites Australia Ltd Buy 1,057 1,076 UBS Scorrites Australia Ltd Buy 1,057 1,092 UBS Scorrites Australia Ltd Buy 1,042 1,092 UBS Scorrites Australia Ltd Buy 1,042 1,092 UBS Scorrites Australia Ltd Buy 2,070 3,071 UBS Scorrites Australia Ltd Buy	JUI-14	UBS Securities Australia Ltd	Buy	7,657	791	Ordinary
UBS Securities Australia Ltd Buy 11,142 1,461 UBS Securities Australia Ltd Buy 5,034 2,50 UBS Securities Australia Ltd Buy 2,034 2,50 UBS Securities Australia Ltd Buy 2,504 2,50 UBS Securities Australia Ltd Buy 2,037 1,381 UBS Securities Australia Ltd Buy 2,038 2,04 UBS Securities Australia Ltd Buy 2,038 2,04 UBS Securities Australia Ltd Buy 2,038 2,04 UBS Securities Australia Ltd Buy 3,037 4,03 UBS Securities Australia Ltd Buy 3,037 3,00 UBS Securities Australia Ltd Buy 3,03 3,00 UBS Securities Australia Ltd Buy 3,01 3,02 3,01 UBS Securities Australia Ltd Buy 3,01 3,02 3,01 UBS Securities Australia Ltd Buy 3,01 3,01 3,02 UBS Securities Australia Ltd Buy 3,01 3,01 3,02	JUI-14	UBS Securities Australia Ltd	Buy	14,646	1,513	Ordinary
UBS Securities Australia Ltd Buy 5,034 5,004 UBS Securities Australia Ltd Buy 5,004 5,004 UBS Securities Australia Ltd Buy 5,004 2,556 264 UBS Securities Australia Ltd Buy 1,299 2,13 UBS Securities Australia Ltd Buy 8,107 4,000 UBS Securities Australia Ltd Buy 8,100 7,402 5,294 UBS Securities Australia Ltd Buy 7,602 5,294 4,124 UBS Securities Australia Ltd Buy 8,100 7,602 5,294 UBS Securities Australia Ltd Buy 7,602 5,294 1,254 UBS Securities Australia Ltd Buy 7,602 5,294 1,254 UBS Securities Australia Ltd	JU-14	UBS Securities Australia Ltd	Buy	11,113	1,148	Ordinary
UBS Securities Australia Ltd Buy 5,034 520 UBS Securities Australia Ltd Buy 2,569 128 UBS Securities Australia Ltd Buy 2,509 134 UBS Securities Australia Ltd Buy 2,007 400 UBS Securities Australia Ltd Buy 2,007 400 UBS Securities Australia Ltd Buy 8,007 8,007 UBS Securities Australia Ltd Buy 8,007 8,000 UBS Securities Australia Ltd Buy 8,007 1,003 UBS Securities Australia Ltd Buy 1,608 1,003 UBS Securities Australia Ltd Buy 1,003 1,003 UBS Securities Australia Ltd Buy 1,004 1,003 UBS Securities Australia Ltd Buy <td>JUI-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>14,142</td> <td>1,461</td> <td>Ordinary</td>	JUI-14	UBS Securities Australia Ltd	Buy	14,142	1,461	Ordinary
UBS Securities Australia Ltd Bluy 1,2556 264 UBS Securities Australia Ltd Bluy 1,250 134 UBS Securities Australia Ltd Bluy 3,872 4,00 UBS Securities Australia Ltd Bluy 2,882 2,594 UBS Securities Australia Ltd Bluy 2,882 2,594 UBS Securities Australia Ltd Bluy 2,892 2,740 1,103 UBS Securities Australia Ltd Bluy 3,894 1,103 4,00 UBS Securities Australia Ltd Bluy 3,90 1,103 4,00 UBS Securities Australia Ltd Bluy 3,00 1,20 1,20 UBS Securities Australia Ltd Bluy 3,00 1,24 1,394 UBS Securities Australia Ltd Bluy 2,00 3,00 1,24 UBS Securities Australia Ltd	Jul- 14	UBS Securities Australia Ltd	Buy	5,034	520	Ordinary
UBS Securities Australia Ltd Bluy 1,297 134 UBS Securities Australia Ltd Bluy 2,062 213 UBS Securities Australia Ltd Bluy 2,062 213 UBS Securities Australia Ltd Bluy 8,727 9,000 UBS Securities Australia Ltd Bluy 8,727 9,000 UBS Securities Australia Ltd Bluy 1,481 1,481 1,481 UBS Securities Australia Ltd Bluy 1,482 1,481 1,481 1,481 1,181 1,481 1,182 1,481 1,182 1,481	Jul-14	UBS Securities Australia Ltd	Buy	2,556	264	Ordinary
UBS Securities Australia Ltd Bluy 3,052 2.03 UBS Securities Australia Ltd Bluy 3,075 400 UBS Securities Australia Ltd Bluy 3,872 400 UBS Securities Australia Ltd Bluy 1,433 1,48 UBS Securities Australia Ltd Bluy 1,892 2,594 UBS Securities Australia Ltd Bluy 2,892 2,594 UBS Securities Australia Ltd Bluy 3,694 1,103 UBS Securities Australia Ltd Bluy 2,740 1,593 UBS Securities Australia Ltd Bluy 2,740 1,594 UBS Securities Australia Ltd Bluy 2,740 1,695 UBS Securities Australia Ltd Bluy 2,740 3,875 4,096 UBS Securities Australia Ltd Bluy 3,071 3,875 4,00 1,24 <td>Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>1,297</td> <td>134</td> <td>Ordinary</td>	Jul-14	UBS Securities Australia Ltd	Buy	1,297	134	Ordinary
UBS Securities Australia Ltd Buy 8 3872 400 UBS Securities Australia Ltd Buy 8 3775 9 000 UBS Securities Australia Ltd Buy 8 3775 9 000 UBS Securities Australia Ltd Buy 1 483 1 481 UBS Securities Australia Ltd Buy 1 482 2 294 UBS Securities Australia Ltd Buy 1 482 2 294 UBS Securities Australia Ltd Buy 1 482 2 294 UBS Securities Australia Ltd Buy 1 482 2 294 3 286 UBS Securities Australia Ltd Buy 1 692 2 294 4 298 UBS Securities Australia Ltd Buy 2 760 1 103 1 280 UBS Securities Australia Ltd Buy 2 710 2 280 2 280 UBS Securities Australia Ltd Buy 2 710 2 280 2 280 UBS Securities Australia Ltd Buy 2 710 2 280 2 280 UBS Securities Australia Ltd Buy 2 710 2 280 2 280 UBS Securitie	Jul-14	UBS Securities Australia Ltd	Buy	2,062	213	Ordinary
UBS Securities Australia Ltd Buy 87.05 9.00 UBS Securities Australia Ltd Buy 87.2 9.00 UBS Securities Australia Ltd Buy 8.82 2.04 UBS Securities Australia Ltd Buy 28.92 2.04 UBS Securities Australia Ltd Buy 28.92 2.094 UBS Securities Australia Ltd Buy 28.92 2.094 UBS Securities Australia Ltd Buy 28.92 2.094 UBS Securities Australia Ltd Buy 3.8544 1.394 UBS Securities Australia Ltd Buy 5.02 7.018 UBS Securities Australia Ltd Buy 5.02 7.02 UBS Securities Australia Ltd Buy 1.05 1.03 UBS Securities Australia Ltd Buy 2.70 3.81 UBS Securities Australia Ltd Buy 2.70 3.82 UBS Securities Australia Ltd Buy 2.70 3.82 UBS Securities Australia Ltd Buy 2.70 3.82 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	3,872	400	Ordinary
UBS Securities Australia Ltd Buy 3872 400 UBS Securities Australia Ltd Buy 3872 401 UBS Securities Australia Ltd Buy 2,982 2,984 UBS Securities Australia Ltd Buy 438 2,018 UBS Securities Australia Ltd Buy 438 2,018 UBS Securities Australia Ltd Buy 19,524 2,018 UBS Securities Australia Ltd Buy 19,524 2,018 UBS Securities Australia Ltd Buy 10,571 1,032 UBS Securities Australia Ltd Buy 5,109 2,808 UBS Securities Australia Ltd Buy 5,109 1,022 UBS Securities Australia Ltd Buy 5,018 1,022 UBS Securities Australia Ltd Buy 5,019 1,022 UBS Securities Australia Ltd Buy 5,019 1,021 UBS Securities Australia Ltd Buy 2,710 2,88 UBS Securities Australia Ltd Buy 2,710 2,88 UBS Securities Australia Ltd Buy <td>Jui-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>87,075</td> <td>000'6</td> <td>Ordinary</td>	Jui-14	UBS Securities Australia Ltd	Buy	87,075	000'6	Ordinary
UBS Securities Australia Ltd Buy 1433 148 UBS Securities Australia Ltd Buy 1,433 1,435 UBS Securities Australia Ltd Buy 28,582 2,994 UBS Securities Australia Ltd Buy 1,503 7,103 UBS Securities Australia Ltd Buy 1,602 7,103 UBS Securities Australia Ltd Buy 7,602 7,933 UBS Securities Australia Ltd Buy 7,602 7,933 UBS Securities Australia Ltd Buy 7,608 7,808 UBS Securities Australia Ltd Buy 7,609 7,808 UBS Securities Australia Ltd Buy 7,009 6,195 6,00 7,808 UBS Securities Australia Ltd Buy 7,009 6,195 6,202 7,308 UBS Securities Australia Ltd Buy 6,195 6,105 7,000 7,000 UBS Securities Australia Ltd Buy 1,219 1,252 2,237 4,238 UBS Securities Australia Ltd Buy 1,210 1,252 2,237	Jul-14	UBS Securities Australia Ltd	Buy	3,872	400	Ordinary
UBS Securities Australia Ltd Bby 1,433 1,483 UBS Securities Australia Ltd Buy 1,922 2,934 UBS Securities Australia Ltd Buy 1,952 2,936 UBS Securities Australia Ltd Buy 6,922 7,13 UBS Securities Australia Ltd Buy 7,608 7,86 UBS Securities Australia Ltd Buy 7,608 7,89 UBS Securities Australia Ltd Buy 7,608 7,89 UBS Securities Australia Ltd Buy 7,608 7,89 UBS Securities Australia Ltd Buy 7,119 1,092 UBS Securities Australia Ltd Buy 7,119 1,252 UBS Securities Australia Ltd Buy 7,608 7,805 UBS Securities Australia Ltd Buy 7,119 1,292 UBS Securities Australia Ltd Buy 2,719 1,292 UBS Securities Australia Ltd Buy 2,735 4,386 UBS Securities Australia Ltd Buy 1,249 1,249 UBS Securities Australia Ltd Bu	Jul-14	UBS Securities Australia Ltd	Buy	188	16	Ordinary
UBS Securities Australia Ltd Buy 28,982 2,994 UBS Securities Australia Ltd Buy 435 2,016 UBS Securities Australia Ltd Buy 19,524 2,016 UBS Securities Australia Ltd Buy 13,494 3,7402 2,016 UBS Securities Australia Ltd Buy 13,494 1,394 40 UBS Securities Australia Ltd Buy 5,602 7,802 1,103 UBS Securities Australia Ltd Buy 5,602 5,933 1,103 UBS Securities Australia Ltd Buy 5,602 5,802 6,802 1,002 UBS Securities Australia Ltd Buy 5,602 5,802 6,802 1,003 1,0	Jul-14	UBS Securities Australia Ltd	Buy	1,433	148	Ordinary
UBS Securities Australia Ltd Bby 435 45 UBS Securities Australia Ltd Bby 435 2018 UBS Securities Australia Ltd Bby 89 396 713 UBS Securities Australia Ltd Bby 89 768 <th< td=""><td>Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>28,982</td><td>2,994</td><td>Ordinary</td></th<>	Jul-14	UBS Securities Australia Ltd	Buy	28,982	2,994	Ordinary
UBS Securities Australia Ltd Bby 19524 2,018 UBS Securities Australia Ltd Bby 1,924 1,134 UBS Securities Australia Ltd Bby 1,144 1,134 UBS Securities Australia Ltd Bby 5,7402 7,88 UBS Securities Australia Ltd Bby 7,688 7,88 UBS Securities Australia Ltd Bby 7,688 7,88 UBS Securities Australia Ltd Bby 6,992 7,688 7,88 UBS Securities Australia Ltd Bby 6,092 7,103 1,03 UBS Securities Australia Ltd Bby 6,098 6,092 7,88 UBS Securities Australia Ltd Bby 6,098 6,092 7,00 2,89 UBS Securities Australia Ltd Bby 1,1249 1,249 1,249 1,249 1,249 UBS Securities Australia Ltd Bby 1,249 1,249 1,249 1,249 1,249 1,249 1,249 1,249 1,249 1,249 1,249 1,249 1,249 1,249 1,24	Jul-14	UBS Securities Australia Ltd	Buy	436	45	Ordinary
UBS Securities Australia Ltd Bby 6902 713 UBS Securities Australia Ltd Bby 38,584 3,986 UBS Securities Australia Ltd Bby 7,402 5,933 UBS Securities Australia Ltd Bby 7,602 7,862 UBS Securities Australia Ltd Bby 7,607 1,103 UBS Securities Australia Ltd Bby 10,677 1,103 UBS Securities Australia Ltd Bby 10,571 1,103 UBS Securities Australia Ltd Bby 2,710 2,88 UBS Securities Australia Ltd Bby 2,710 1,022 UBS Securities Australia Ltd Bby 5,634 5,83 UBS Securities Australia Ltd Bby 1,016 1,05 UBS Securities Australia Ltd Bby 3,117 3,22 UBS Securities Australia Ltd Bby 3,10 3,11 UBS Securities Australia Ltd Bby 3,10 3,11 UBS Securities Australia Ltd Bby 3,20 1,28 UBS Securities Australia Ltd Bby <td>Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>19,524</td> <td>2,018</td> <td>Ordinary</td>	Jul-14	UBS Securities Australia Ltd	Buy	19,524	2,018	Ordinary
UBS Securities Australia Ltd Buy 38,584 3,986 UBS Securities Australia Ltd Buy 57,402 5,933 UBS Securities Australia Ltd Buy 7,602 5,933 UBS Securities Australia Ltd Buy 7,602 7,60 UBS Securities Australia Ltd Buy 6,193 40 UBS Securities Australia Ltd Buy 6,193 6,09 UBS Securities Australia Ltd Buy 1,103 1,222 UBS Securities Australia Ltd Buy 1,103 1,282 UBS Securities Australia Ltd Buy 1,016 1,082 UBS Securities Australia Ltd Buy 5,692 5,88 UBS Securities Australia Ltd Buy 1,016 1,08 UBS Securities Australia Ltd Buy 4,2,435 4,386 UBS Securities Australia Ltd Buy 1,258 1,436 UBS Securities Australia Ltd Buy 1,243 1,43 UBS Securities Australia Ltd Buy 2,715 1,28 UBS Securities Australia Ltd Bu	Jul-14	UBS Securities Australia Ltd	Buy	6,902	713	Ordinary
UBS Securites Australia Ltd Buy 13,494 1,394 UBS Securites Australia Ltd Buy 7,602 5,933 UBS Securites Australia Ltd Buy 7,608 7,605 UBS Securities Australia Ltd Buy 10,571 1,103 UBS Securities Australia Ltd Buy 30,124 3,112 UBS Securities Australia Ltd Buy 30,124 3,112 UBS Securities Australia Ltd Buy 3,171 1,082 UBS Securities Australia Ltd Buy 2,710 2,80 UBS Securities Australia Ltd Buy 2,710 2,80 UBS Securities Australia Ltd Buy 2,710 1,249 UBS Securities Australia Ltd Buy 1,249 1,249 UBS Securities Australia Ltd Buy 1,249 1,24 UBS Securities Australia Ltd Buy 2,525 2,327 UBS Securities Australia Ltd Buy 2,525 2,327 UBS Securities Australia Ltd Buy 2,525 2,525 UBS Securities Australia Ltd <td< td=""><td>Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>38.584</td><td>3.986</td><td>Ordinary</td></td<>	Jul-14	UBS Securities Australia Ltd	Buy	38.584	3.986	Ordinary
UBS Securities Australia Ltd Buy 57,402 5,933 UBS Securities Australia Ltd Buy 10,677 1,103 UBS Securities Australia Ltd Buy 6,195 6,40 UBS Securities Australia Ltd Buy 6,195 6,192 UBS Securities Australia Ltd Buy 10,571 1,1092 UBS Securities Australia Ltd Buy 2,710 2,88 UBS Securities Australia Ltd Buy 5,692 5,88 UBS Securities Australia Ltd Buy 1,249 1,24 UBS Securities Australia Ltd Buy 2,732 1,43 UBS Securities Australia Ltd Buy 2,562 2,37 UBS Securities Australia Ltd Buy 2,562 2,37 UBS Securities Australia Ltd Buy 2,562 2,37 UBS Securities Australia Ltd Buy<	Jul-14	UBS Securities Australia Ltd	Buy	13,494	1.394	Ordinary
UBS Securities Australia Ltd Buy 1,608 UBS Securities Australia Ltd Buy 1,608 UBS Securities Australia Ltd Buy 6,193 UBS Securities Australia Ltd Buy 30,124 UBS Securities Australia Ltd Buy 30,124 UBS Securities Australia Ltd Buy 30,124 UBS Securities Australia Ltd Buy 1,2719 UBS Securities Australia Ltd Buy 1,2719 UBS Securities Australia Ltd Buy 1,2719 UBS Securities Australia Ltd Buy 5,622 UBS Securities Australia Ltd Buy 1,249 UBS Securities Australia Ltd Buy 1,344 UBS Securities Australia Ltd Buy 1,344 UBS Securities Australia Ltd Buy 1,349 UBS Securities Australia Ltd Buy 1,344 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buv	57.402	5 933	Ordinary
UBS Securities Australia Ltd Buy 1,103 UBS Securities Australia Ltd Buy 6,195 640 UBS Securities Australia Ltd Buy 38,7 40 UBS Securities Australia Ltd Buy 30,124 3,112 UBS Securities Australia Ltd Buy 2,710 2,80 UBS Securities Australia Ltd Buy 2,710 2,80 UBS Securities Australia Ltd Buy 2,710 3,22 UBS Securities Australia Ltd Buy 2,710 3,22 UBS Securities Australia Ltd Buy 1,249 1,26 UBS Securities Australia Ltd Buy 1,249 1,26 UBS Securities Australia Ltd Buy 1,249 1,24 UBS Securities Australia Ltd Buy 1,243 4,386 UBS Securities Australia Ltd Buy 1,243 1,44 UBS Securities Australia Ltd Buy 2,525 2,327 UBS Securities Australia Ltd Buy 3,00 3,14 UBS Securities Australia Ltd Buy 3,00	Jul-14	UBS Securities Australia Ltd	Buv	7.608	786	Ordinary
UBS Securities Australia Ltd Buy 6,195 6,09 UBS Securities Australia Ltd Buy 6,195 6,195 UBS Securities Australia Ltd Buy 10,271 1,102 UBS Securities Australia Ltd Buy 10,571 1,082 UBS Securities Australia Ltd Buy 5,692 5,89 UBS Securities Australia Ltd Buy 5,692 5,89 UBS Securities Australia Ltd Buy 1,016 105 UBS Securities Australia Ltd Buy 1,249 105 UBS Securities Australia Ltd Buy 1,249 105 UBS Securities Australia Ltd Buy 1,249 1,44 UBS Securities Australia Ltd Buy 1,250 1,44 UBS Securities Australia Ltd Buy 2,552 2,327 UBS Securities Australia Ltd Buy 1,394 1,44 UBS Securities Australia Ltd Buy 3,002 3,707 UBS Securities Australia Ltd Buy 3,000 3,134 UBS Securities Australia Ltd Buy <td>Jul-14</td> <td>UBS Securities Australia Itd</td> <td>8.10</td> <td>10,677</td> <td>1 103</td> <td>Ordinary</td>	Jul-14	UBS Securities Australia Itd	8.10	10,677	1 103	Ordinary
USS Securities Australia Ltd Buy 38.7 4.0 USS Securities Australia Ltd Buy 30.124 3.112 USS Securities Australia Ltd Buy 10,571 1.092 USS Securities Australia Ltd Buy 1,2719 1.282 USS Securities Australia Ltd Buy 1,2719 1.282 USS Securities Australia Ltd Buy 1,016 1.05 USS Securities Australia Ltd Buy 1,016 1.05 USS Securities Australia Ltd Buy 1,249 1.24 USS Securities Australia Ltd Buy 1,249 1.36 USS Securities Australia Ltd Buy 1,249 1.36 USS Securities Australia Ltd Buy 1,249 1.36 USS Securities Australia Ltd Buy 2,525 2,327 USS Securities Australia Ltd Buy 1,540 1.36 USS Securities Australia Ltd Buy 1,510 1.51 USS Securities Australia Ltd Buy 1,520 2,327 USS Securities Australia Ltd Buy<	11-14	I IBS Sequities Australia I td	200	£ 10E	040	Ordinary.
UBS Securities Australia Ltd Buy 30,124 3,12 UBS Securities Australia Ltd Buy 10,571 1,092 UBS Securities Australia Ltd Buy 2,710 280 UBS Securities Australia Ltd Buy 2,710 280 UBS Securities Australia Ltd Buy 2,710 280 UBS Securities Australia Ltd Buy 2,710 1,28 UBS Securities Australia Ltd Buy 1,016 1,05 UBS Securities Australia Ltd Buy 1,243 1,00 UBS Securities Australia Ltd Buy 1,243 1,00 UBS Securities Australia Ltd Buy 1,258 1,30 UBS Securities Australia Ltd Buy 2,532 2,327 UBS Securities Australia Ltd Buy 2,558 1,44 UBS Securities Australia Ltd Buy 2,558 1,44 UBS Securities Australia Ltd Buy 2,558 1,44 UBS Securities Australia Ltd Buy 3,000 3,13 UBS Securities Australia Ltd Buy	11-17	LIBC Counties Australia 1+d	600	261,0	2 5	Cidinaly
UBS Securities Australia Ltd Buy 10,571 1,022 UBS Securities Australia Ltd Buy 10,571 1,022 UBS Securities Australia Ltd Buy 2,710 280 UBS Securities Australia Ltd Buy 5,692 588 UBS Securities Australia Ltd Buy 1,016 105 UBS Securities Australia Ltd Buy 3,301 341 UBS Securities Australia Ltd Buy 1,249 1,29 UBS Securities Australia Ltd Buy 3,672 400 UBS Securities Australia Ltd Buy 1,249 1,44 UBS Securities Australia Ltd Buy 1,256 1,39 UBS Securities Australia Ltd Buy 1,256 1,44 UBS Securities Australia Ltd Buy 1,594 1,44 UBS Securities Australia Ltd Buy 1,506 1,504 UBS Securities Australia Ltd Buy 3,003 3,13 UBS Securities Australia Ltd Buy 3,000 3,13 UBS Securities Australia Ltd Buy	1-1-1	I IRS Securities Australia 14d	620	20 124	2	Ordinary
UBS Securities Australia Ltd Buy 10,371 1,092 UBS Securities Australia Ltd Buy 2,710 120 UBS Securities Australia Ltd Buy 2,710 280 UBS Securities Australia Ltd Buy 5,038 630 UBS Securities Australia Ltd Buy 1,016 105 UBS Securities Australia Ltd Buy 3,301 341 UBS Securities Australia Ltd Buy 42,435 4,386 UBS Securities Australia Ltd Buy 42,435 4,386 UBS Securities Australia Ltd Buy 2,527 400 UBS Securities Australia Ltd Buy 2,525 2,327 UBS Securities Australia Ltd Buy 2,525 2,327 UBS Securities Australia Ltd Buy 1,510 1,685 UBS Securities Australia Ltd Buy 1,520 1,685 UBS Securities Australia Ltd Buy 1,530 3,134 UBS Securities Australia Ltd Buy 1,530 3,144 UBS Securities Australia Ltd Buy <td></td> <td>OBS Securities Australia Ltd</td> <td>bag</td> <td>30,124</td> <td>2,112</td> <td>Ordinary</td>		OBS Securities Australia Ltd	bag	30,124	2,112	Ordinary
UBS Securities Australia Ltd Buy 1,252 UBS Securities Australia Ltd Buy 2,719 2,88 UBS Securities Australia Ltd Buy 6,098 6,692 6,88 UBS Securities Australia Ltd Buy 1,016 10,59 3,117 3,22 UBS Securities Australia Ltd Buy 3,117 3,22 4,386 UBS Securities Australia Ltd Buy 4,243 4,386 UBS Securities Australia Ltd Buy 4,243 4,386 UBS Securities Australia Ltd Buy 2,552 1,397 UBS Securities Australia Ltd Buy 2,525 1,396 UBS Securities Australia Ltd Buy 2,554 1,44 UBS Securities Australia Ltd Buy 2,554 1,44 UBS Securities Australia Ltd Buy 2,554 1,44 UBS Securities Australia Ltd Buy 2,554 1,510 1,55 UBS Securities Australia Ltd Buy 3,020 3,433 UBS Securities Australia Ltd Buy 3,502 1		Obs Securities Australia Ltd	gnà	10,5/01	760,1	Ordinary
UBS Securities Australia Ltd Buy 5,692 580 UBS Securities Australia Ltd Buy 6,098 699 <	Jul-14	UBS Securities Australia Ltd	Buy	12,119	1,252	Ordinary
UBS Securities Australia Ltd Buy 6,098 680 UBS Securities Australia Ltd Buy 5,632 588 UBS Securities Australia Ltd Buy 1,016 1,03 UBS Securities Australia Ltd Buy 1,249 1,249 UBS Securities Australia Ltd Buy 4,2435 4,386 UBS Securities Australia Ltd Buy 2,2435 4,386 UBS Securities Australia Ltd Buy 2,534 5,83 UBS Securities Australia Ltd Buy 2,525 2,327 UBS Securities Australia Ltd Buy 1,510 1,685 UBS Securities Australia Ltd Buy 3,030 3,13 UBS Securities Australia Ltd Buy 3,030 3,13 UBS Securities Australia Ltd Buy 3,030 3,13 UBS Securities Australia Ltd Buy<	Jul-14	UBS Securities Australia Ltd	Buy	2,710	280	Ordinary
UBS Securities Australia Ltd Buy 5,692 588 UBS Securities Australia Ltd Buy 1,016 1,05 UBS Securities Australia Ltd Buy 3,301 341 UBS Securities Australia Ltd Buy 42,435 4,286 UBS Securities Australia Ltd Buy 42,435 4,286 UBS Securities Australia Ltd Buy 42,435 1,286 UBS Securities Australia Ltd Buy 1,259 1,144 UBS Securities Australia Ltd Buy 25,525 2,327 UBS Securities Australia Ltd Buy 2,525 2,327 UBS Securities Australia Ltd Buy 1,510 1,56 UBS Securities Australia Ltd Buy 1,510 1,56 UBS Securities Australia Ltd Buy 1,510 1,510 UBS Securities Australia Ltd Buy 1,510 1,510 UBS Securities Australia Ltd Buy 1,510 1,510 UBS Securities Australia Ltd Buy 3,502 3,483 UBS Securities Australia Ltd	Jul-14	UBS Securities Australia Ltd	Buy	860'9	630	Ordinary
UBS Securities Australia Ltd Buy 1,016 105 UBS Securities Australia Ltd Buy 3,117 3.22 UBS Securities Australia Ltd Buy 1,249 1.79 UBS Securities Australia Ltd Buy 42,435 4,000 UBS Securities Australia Ltd Buy 5,634 582 UBS Securities Australia Ltd Buy 1,256 130 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 1,510 116 UBS Securities Australia Ltd Buy 1,510 1,56 UBS Securities Australia Ltd Buy 1,510 1,56 UBS Securities Australia Ltd Buy 1,510 1,56 UBS Securities Australia Ltd Buy 1,510 1,510 UBS Securities Australia Ltd Buy 1,510 1,510 UBS Securities Australia Ltd Buy 1,520 1,514 UBS Securities Australia Ltd Buy 1,500 1,174 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	5,692	288	Ordinary
UBS Securities Australia Ltd Buy 3,117 322 UBS Securities Australia Ltd Buy 1,249 129 UBS Securities Australia Ltd Buy 4,2435 4,386 UBS Securities Australia Ltd Buy 4,2435 4,386 UBS Securities Australia Ltd Buy 5,634 582 UBS Securities Australia Ltd Buy 1,256 130 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 3,030 313 UBS Securities Australia Ltd Buy 1,510 1,655 UBS Securities Australia Ltd Buy 3,030 313 UBS Securities Australia Ltd Buy 3,030 313 UBS Securities Australia Ltd Buy 3,030 3,63 UBS Securities Australia Ltd Buy 3,030 3,63 UBS Securities Australia Ltd Buy 3,030 3,63 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	1,016	105	Ordinary
UBS Securities Australia Ltd Buy 3,301 341 UBS Securities Australia Ltd Buy 42,435 436 UBS Securities Australia Ltd Buy 3,672 400 UBS Securities Australia Ltd Buy 1,258 134 UBS Securities Australia Ltd Buy 1,258 134 UBS Securities Australia Ltd Buy 2,525 2,327 UBS Securities Australia Ltd Buy 3,707 383 UBS Securities Australia Ltd Buy 3,707 383 UBS Securities Australia Ltd Buy 1,510 1,66 UBS Securities Australia Ltd Buy 1,500 1,685 UBS Securities Australia Ltd Buy 1,500 1,685 UBS Securities Australia Ltd Buy 1,500 1,714 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	3.117	322	Ordinary
UBS Securities Australia Ltd Buy 1,249 1,29 UBS Securities Australia Ltd Buy 4,2435 4,386 UBS Securities Australia Ltd Buy 5,634 4,386 UBS Securities Australia Ltd Buy 1,258 130 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 1,510 1,65 UBS Securities Australia Ltd Buy 1,510 1,65 UBS Securities Australia Ltd Buy 1,530 1,14 UBS Securities Australia Ltd Buy 1,530 1,685 UBS Securities Australia Ltd Buy 1,394 1,44 UBS Securities Australia Ltd Buy 3,002 3,13 UBS Securities Australia Ltd Buy 3,000 3,13 UBS Securities Australia Ltd Buy 3,000 3,13 UBS Securities Australia Ltd Buy 3,000 3,001 UBS Securities Australia Ltd Buy<	Jul-14	UBS Securities Australia Hd	Biv	3 301	341	Ordinary
UBS Securities Australia Ltd Buy 42,435 4,145 UBS Securities Australia Ltd Buy 42,435 4,145 UBS Securities Australia Ltd Buy 1,258 130 UBS Securities Australia Ltd Buy 1,394 144 UBS Securities Australia Ltd Buy 3,030 313 UBS Securities Australia Ltd Buy 3,030 313 UBS Securities Australia Ltd Buy 1,510 1,56 UBS Securities Australia Ltd Buy 1,510 1,58 UBS Securities Australia Ltd Buy 1,530 1,685 UBS Securities Australia Ltd Buy 1,540 1,540 UBS Securities Australia Ltd Buy 1,540 1,540 UBS Securities Australia Ltd Buy 3,508 3,483 UBS Securities Australia Ltd Buy 3,508 3,483 UBS Securities Australia Ltd Buy 3,508 3,483 UBS Securities Australia Ltd Buy 3,508 3,608 UBS Securities Australia Ltd Buy </td <td>11-14</td> <td>I IBS Securities Australia I to</td> <td>82)</td> <td>1 240</td> <td>120</td> <td>Ordinary</td>	11-14	I IBS Securities Australia I to	82)	1 240	120	Ordinary
UBS Securities Australia Ltd Buy 4,2435 4,380 UBS Securities Australia Ltd Buy 5,634 582 UBS Securities Australia Ltd Buy 1,558 130 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 1,510 156 UBS Securities Australia Ltd Buy 1,510 156 UBS Securities Australia Ltd Buy 1,510 1,510 UBS Securities Australia Ltd Buy 1,520 1,685 UBS Securities Australia Ltd Buy 1,500 3,134 UBS Securities Australia Ltd Buy 3,028 3,134 UBS Securities Australia Ltd Buy 3,028 3,124 UBS Securities Australia Ltd Buy 3,028 3,124 UBS Securities Australia Ltd Buy 1,0 1,134 UBS Securities Australia Ltd Buy 1,0 1,134 UBS Securities Australia Ltd Buy <td></td> <td>ODS Securities Australia Ltd</td> <td>had</td> <td>242,1</td> <td>671</td> <td>Ordinary</td>		ODS Securities Australia Ltd	had	242,1	671	Ordinary
UBS Securities Australia Ltd Buy 5.832 400 UBS Securities Australia Ltd Buy 5.634 582 UBS Securities Australia Ltd Buy 1,258 130 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 3,030 313 UBS Securities Australia Ltd Buy 1,394 144 UBS Securities Australia Ltd Buy 1,394 144 UBS Securities Australia Ltd Buy 1,394 144 UBS Securities Australia Ltd Buy 3,028 318 UBS Securities Australia Ltd Buy 49,600 5,14 UBS Securities Australia Ltd Buy 49,600 5,14 UBS Securities Australia Ltd Buy 35,028 3,483 UBS Securities Australia Ltd Buy 35,028 3,483 UBS Securities Australia Ltd Buy 10,000 5,174 UBS Securities Australia Ltd Buy 182,940 18,713 UBS Securities Australia Ltd Buy <td>Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>42,435</td> <td>4,386</td> <td>Ordinary</td>	Jul-14	UBS Securities Australia Ltd	Buy	42,435	4,386	Ordinary
UBS Securities Australia Ltd Buy 1,258 190 UBS Securities Australia Ltd Buy 1,258 134 UBS Securities Australia Ltd Buy 1,394 144 UBS Securities Australia Ltd Buy 3,707 383 UBS Securities Australia Ltd Buy 3,030 313 UBS Securities Australia Ltd Buy 1,510 1,685 UBS Securities Australia Ltd Buy 1,394 1,44 UBS Securities Australia Ltd Buy 3,028 313 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 33,505 3,483 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 10 1,134 UBS Securities Australia Ltd Buy 1,346 1,366 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	3,872	400	Ordinary
UBS Securities Australia Ltd Buy 1,258 130 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 3,707 383 UBS Securities Australia Ltd Buy 1,510 1,655 UBS Securities Australia Ltd Buy 1,530 1,685 UBS Securities Australia Ltd Buy 49,600 5,174 UBS Securities Australia Ltd Buy 49,600 5,174 UBS Securities Australia Ltd Buy 33,505 3,635 UBS Securities Australia Ltd Buy 33,505 1,714 UBS Securities Australia Ltd Buy 1,80 1,714 UBS Securities Australia Ltd Buy 1,80 1,87 UBS Securities Australia Ltd Such 1,80 1,87 UBS Securities Australia Ltd Buy 1,80 1,87 UBS Securities Australia Ltd Buy 1,80 1,87 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	5,634	285	Ordinary
UBS Securities Australia Ltd Buy 1,394 144 UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 3,030 313 UBS Securities Australia Ltd Buy 1,510 1,56 UBS Securities Australia Ltd Buy 1,530 1,685 UBS Securities Australia Ltd Buy 1,530 1,685 UBS Securities Australia Ltd Buy 1,394 1,144 UBS Securities Australia Ltd Buy 4,9500 5,124 UBS Securities Australia Ltd Buy 33,505 3,483 UBS Securities Australia Ltd Buy 3,600 5,124 UBS Securities Australia Ltd Buy 1,82,675 (18,713) UBS Securities Australia Ltd Stock returned N/A (8,253) UBS Securities Australia Ltd Stock returned N/A (1,895) UBS Securities Australia Ltd Buy 1,378 1,517 UBS Securities Australia Ltd Buy 1,378 1,517 UBS Securities Au	Jul-14	UBS Securities Australia Ltd	Buy	1,258	130	Ordinary
UBS Securities Australia Ltd Buy 22,525 2,327 UBS Securities Australia Ltd Buy 3,707 383 UBS Securities Australia Ltd Buy 1,510 1.65 UBS Securities Australia Ltd Buy 1,520 1,685 UBS Securities Australia Ltd Buy 1,324 1,44 UBS Securities Australia Ltd Buy 4,6600 5,124 UBS Securities Australia Ltd Buy 4,6600 5,124 UBS Securities Australia Ltd Buy 4,6600 5,124 UBS Securities Australia Ltd Buy 1,0 1 UBS Securities Australia Ltd Buy 1,0 1 UBS Securities Australia Ltd Buy 1,0 1 UBS Securities Australia Ltd Stock returned NA (8,173) UBS Securities Australia Ltd Stock returned NA (8,233) UBS Securities Australia Ltd Buy 29,730 3,042 UBS Securities Australia Ltd Buy 1,378 1,577 UBS Securities Australia Ltd	Jul-14	UBS Securities Australia Ltd	Buv	1.394	144	Ordinary
UBS Securities Australia Ltd Buy 3,020 318 UBS Securities Australia Ltd Buy 1,510 156 UBS Securities Australia Ltd Buy 1,394 144 UBS Securities Australia Ltd Buy 1,394 1,44 UBS Securities Australia Ltd Buy 3,028 313 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 35,055 3,463 UBS Securities Australia Ltd Buy 35,055 3,463 UBS Securities Australia Ltd Buy 35,055 3,463 UBS Securities Australia Ltd Buy 10,713 10 UBS Securities Australia Ltd Stock returned NA (18,739) UBS Securities Australia Ltd Stock returned NA (2,1895) UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia	11-14	LIBS Securities Australia Ltd	Biry	22 525	2 327	Ordinan
UBS Securities Australia Ltd Buy 3,707 363 UBS Securities Australia Ltd Buy 1,510 116 UBS Securities Australia Ltd Buy 1,510 1,685 UBS Securities Australia Ltd Buy 1,324 1,144 UBS Securities Australia Ltd Buy 4,6600 5,124 UBS Securities Australia Ltd Buy 1,0 1,0 UBS Securities Australia Ltd Buy 1,0 1,1 UBS Securities Australia Ltd Buy 1,8,7,94 1,8,7,13 UBS Securities Australia Ltd Stock returned N/A 18,7,13 UBS Securities Australia Ltd Buy 2,9,730 3,042 UBS Securities Australia Ltd Buy 1,3,789 1,577 UBS Securities Australia Ltd Buy 1,3,789 1,577 UBS Securities Australia Ltd	177	Late Constitution Australia Late	(cond	2020	720,7	Ordinary
UBS Securities Australia Ltd Buy 1,510 1685 UBS Securities Australia Ltd Buy 1,510 1,685 UBS Securities Australia Ltd Buy 1,394 144 UBS Securities Australia Ltd Buy 3,028 313 UBS Securities Australia Ltd Buy 49,600 5,174 UBS Securities Australia Ltd Buy 49,600 5,174 UBS Securities Australia Ltd Buy 10 1 UBS AG London Branch Sell 182,670 187,713 UBS AG London Branch Sell 182,670 187,713 UBS Securities Australia Ltd Stock returned N/A (21,895) UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 13,346 1,517 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd	1 1	COD Secondines Australia Ltd	éna 6	3,707	200	Ordinary
UBS Securities Australia Ltd Buy 1,510 156 UBS Securities Australia Ltd Buy 1,530 1,685 UBS Securities Australia Ltd Buy 1,394 1,44 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 35,025 3,463 UBS Securities Australia Ltd Buy 35,055 3,463 UBS Securities Australia Ltd Buy 10,713 10 UBS Securities Australia Ltd Stock returned N/A (8,253) UBS Securities Australia Ltd Stock returned N/A (8,253) UBS Securities Australia Ltd Stock returned N/A (1,895) UBS Securities Australia Ltd Buy 1,378 1,517 UBS Securities Australia Ltd Buy 1,8789 1,517 UBS Securit	10-14	Opp securities Australia Ltd	buy	3,050	313	Ordinary
UBS Securities Australia Ltd Buy 1,685 UBS Securities Australia Ltd Buy 1,394 1,44 UBS Securities Australia Ltd Buy 3,028 3,134 UBS Securities Australia Ltd Buy 4,96,600 5,124 UBS Securities Australia Ltd Buy 33,500 3,463 UBS Securities Australia Ltd Buy 10 1 UBS Securities Australia Ltd Buy 10 1 UBS Securities Australia Ltd Sell 182,73 18,713 UBS Securities Australia Ltd Stock returned NA (8,233) UBS Securities Australia Ltd Buy 29,730 3,042 UBS Securities Australia Ltd Buy 13,346 1,576 UBS Securities Australia Ltd Buy 13,786 1,577 UBS Securities Australia Ltd Buy 18,249 1,577 UBS Securities Australia Ltd Buy 18,240 1,577 UBS Securities Australia Ltd Buy 18,240 1,8713	10I-14	UBS Securities Australia Ltd	Buy	015,1	156	Ordinary
UBS Securities Australia Ltd Bby 1,394 144 UBS Securities Australia Ltd Buy 3,028 313 UBS Securities Australia Ltd Buy 49,600 5,174 UBS Securities Australia Ltd Buy 10 1 UBS Securities Australia Ltd Buy 10 1 UBS AG London Branch Sell 182,670 187,173 UBS AG London Branch Buy 182,670 187,773 UBS AG London Branch Buy 182,670 187,773 UBS Securities Australia Ltd Stock returned N/A (21,895) UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy <td>Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>16,302</td> <td>1,685</td> <td>Ordinary</td>	Jul-14	UBS Securities Australia Ltd	Buy	16,302	1,685	Ordinary
UBS Securities Australia Ltd Buy 3,028 313 UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 10 1 UBS Securities Australia Ltd Buy 10 1 UBS Securities Australia Ltd Self 182,675 (18,713) UBS AG London Branch Buy 182,675 (18,713) UBS AG London Branch Buy 182,675 (18,713) UBS Securities Australia Ltd Stock returned N/A (8,253) UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd Buy 18,249 18,719	Jul-14	UBS Securities Australia Ltd	Buy	1,394	144	Ordinary
UBS Securities Australia Ltd Buy 49,600 5,124 UBS Securities Australia Ltd Buy 33,505 3,463 UBS Securities Australia Ltd Buy 10 1,1 UBS AG London Branch Sell 182,675 (18,713) UBS AG London Branch Buy 182,675 (18,713) UBS AG London Branch Buy (18,713) UBS Securities Australia Ltd Stock returned N/A (21,895) UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 13,746 1,517 UBS Securities Australia Ltd Buy 13,746 1,517 UBS Securities Australia Ltd Buy 1,374 1,517 UBS Securities Australia Ltd Buy 1,374 1,517 UBS Securities Australia Ltd Buy 1,374 1,517 UBS Securities Australia Ltd Buy 1,378 1,517 UBS Securities Australia Ltd Buy 18,249 1,517 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	3,028	313	Ordinary
UBS Securities Australia Ltd Buy 33,505 3,463 UBS Securities Australia Ltd Buy 10 1 UBS AG London Branch Buy 182,675 (18,713) UBS AG London Branch Buy 182,940 18,713 UBS AG London Branch Buy 182,940 18,713 UBS Securities Australia Ltd Stock returned NA (8,223) UBS Securities Australia Ltd Buy 29,730 3,042 UBS Securities Australia Ltd Buy 13,346 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,294 1,517 UBS Securities Australia Ltd Buy 18,294 1,517	Jul-14	UBS Securities Australia Ltd	Buy	49.600	5 174	Ordinary
USS Securities Australia Ltd Boy 105 USS Acturities Australia Ltd Sell 105 USS AC London Branch Sell 182,575 USS AC London Branch Boy 182,940 USS Securities Australia Ltd Stock returned NA (8,233) USS Securities Australia Ltd Boy 29,730 3,042 USS Securities Australia Ltd Buy 13,346 1,576 UBS Securities Australia Ltd Buy 14,789 1,577 UBS Securities Australia Ltd Buy 18,249 1,577 UBS Securities Australia Ltd Buy 18,240 18,713	1.1.14	I IBS Securities Australia 14d	Bis	33 505	2 463	Ordinan,
USS AG London Branch Seq 182,675 (18,713) USS AG London Branch Seq 182,675 (18,713) USS AG London Branch Buy 182,940 18,713 USS AG London Branch Stock returned N/A (8,253) USS Securities Australia Ltd Buy 29,730 3,042 USS Securities Australia Ltd Buy 13,346 1,566 UBS Securities Australia Ltd Buy 13,346 1,517 UBS Securities Australia Ltd Buy 13,346 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,294 1,517	h-1.4	LIBS Securities Australia 14d	Page 1	000	507,0	Ordinary
USS AG London Ranch Set 182,673 UBS AG London Ranch BUY 182,694 18,713 UBS AG London Ranch BUY 182,640 18,713 UBS Securities Australia Ltd Stock returned NA (2,895) UBS Securities Australia Ltd Buy 29,730 3,042 UBS Securities Australia Ltd Buy 13,789 1,517 UBS Securities Australia Ltd Buy 15,77 15,77 UBS Securities Australia Ltd Buy 18,789 1,517		I lot AC London Drongh	(Sell	100 635	- 10077	o diament
UBS Securities Australia Ltd Stock returned NA (8,133) UBS Securities Australia Ltd Stock returned NA (2,1895) UBS Securities Australia Ltd Buy 29,730 3,042 UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 14,789 1,577 UBS Securities Australia Ltd Buy 14,789 1,577 UBS Securities Australia Ltd Buy 18,2940 18,719 UBS Securities Australia Ltd Buy 18,2940 18,719	1 1 1	OBS AG LOI IOU BIRING	liac l	182,675	(18,713)	Orginary
UBS Securities Australia Ltd Stock returned NAA (8,253) UBS Securities Australia Ltd Stock returned NA (21,895) UBS Securities Australia Ltd Buy 23,730 3,042 UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 14,789 1,517 UBS Securities Australia Ltd Buy 1,517 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517 UBS Securities Australia Ltd Buy 18,789 1,517	Jul-14	UBS AG London Branch	Buy	182,940	18,713	Ordinary
UBS Securities Australia Ltd Stock returned NAA (21,895) UBS Securities Australia Ltd Buy 29,730 3,042 UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 14,789 1,517 UBS Securities Australia Ltd Seli 379,047 (38,719) UBS Securities Australia Ltd Buy 18,249 1,517	Jul-14	UBS Securities Australia Ltd	Stock returned	NA	(8,253)	Ordinary
UBS Securities Australia Ltd Buy 29,730 3,042 UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 14,789 1,517 UBS Securities Australia Ltd Sell 37,9047 (38,719) UBS Securities Australia Ltd Buy 18,294 18,719	Jul-14	UBS Securities Australia Ltd	Stock returned	NA	(21,895)	Ordinary
UBS Securities Australia Ltd Buy 13,346 1,366 UBS Securities Australia Ltd Buy 14,789 1,517 UBS Securities Australia Ltd Sell 379,047 (38,719) UBS Securities Australia Ltd Buy 182,940 18,713	Jul-14	UBS Securities Australia Ltd	Buy	29,730	3,042	Ordinary
UBS Securities Australia Ltd Buy 14,789 1,517 UBS Securities Australia Ltd Seli 379,047 (38,719) UBS Securities Australia Ltd Buy 182,940 18,713	Iul-14	UBS Securities Australia Ltd	Buv	13.346	1.366	Ordinary
UBS Securities Australia Ltd Sell 379,047 (38,719) UBS Securities Australia Ltd Buy 182,940 18,713	101-14	UBS Securities Australia Ltd	Buy	14 789	1 517	Ordinary
UBS Securities Australia Ltd Buy 182,940 18,713	11.14	I IBS Securities Australia 14d	la S	220,075	(00 710)	Ordingry
UBS SECURITIES AUSTRAINA LTD 18,713		COO Securities Australia Liu	Joe II	10000		
	V				1-1.100	Çığımı'y

UBS Securities Australia Ltd	Buy	859 25	5 444	Ordinary
UBS Securities Australia Ltd	Buy	13,358	1,380	Ordinary
UBS Securities Australia Ltd	Buy	17,792	1,838	Ordinary
UBS Securities Australia Ltd	Buv	5,498	568	Ordinary
IBS Securities Australia I td	Buy	3,020	312	Ordinary
DO Constitut Australia 14	7.00	2 001	000	Ording
Day of the Arrest of the Arres	600	100,11	200	, and
Securities Australia Ltd	, and	001,01	606,1	Couriery
UBS Securifies Australia Ltd	any.	*	70	Orainary
Securities Australia Ltd	Buy	136	14	Ordinary
UBS Securities Australia Ltd	Buy	21,083	2,178	Ordinary
UBS Securities Australia Ltd	Buv	18,218	1,882	Ordinary
UBS Securities Australia Ltd	Buv	20.318	2.099	Ordinary
Committee Australia 14d	/ 'a	10,590	700	Ordinan
CO Securities Australia Lite	600	12,00	160/-	Viginal of
S Securities Australia Ltd	Buy	1714	171	Ordinary
UBS Securities Australia Ltd	Buy	080'85	6,000	Ordinary
JBS Securities Australia Ltd	Buy	58,080	9,000	Ordinary
UBS Securities Australia Ltd	Buy	28,080	000'9	Ordinary
IBS Securities Australia 1td	Buy	17.714	1.830	Ordinary
Commission American Lead	75.0	1000	000	Granis
CO Seculities Australia Ltd	Buy	150,6	250	Colorinary
S Securities Australia Ltd	Buy	3,049	315	Ordinary
UBS Securities Australia Ltd	Buy	91	2	Ordinary
S Securities Australia Ltd	Buy	10,522	1,087	Ordinary
18S Societies Australia 14d	Blix	19 476	2,012	Ordinan
Continue Automotive Continue C	7.0	31 133	1010	
S Securities Australia Liu	Buy	221,12	70177	dali lary
S Securities Australia Ltd	Buy	18,/11	1,933	Ordinary
UBS Securities Australia Ltd	Buy	4,046	418	Ordinary
UBS Securities Australia Ltd	Buy	57,519	5,942	Ordinary
S Securities Australia I td	Mary	82.280	8.500	Ordinary
IBC Countities Australia Ltd	Bin	47 693	4 927	Ordinary
Secondica Australia etc	600	350 34	720,7	Ordinary
DDS Securities Australia Ltu	ed.	0/7'04	00/14	Olumbia
S Securities Australia Ltd	Buy	41,370	4,2/6	Ordinary
UBS Securities Australia Ltd	Buy	6,873	710	Ordinary
JBS Securities Australia Ltd	Buy	46,077	4,760	Ordinary
JBS Securities Australia Ltd	Buv	10.212	1.055	Ordinary
IBS Sequities Australia I to	Buy	6.873	710	Ordinary
DC Con without Australia 14d	a	2 877	700	Ordina
Control of the Contro	0	4 25.4	00,	
UBS Securities Australia LIG	hng	4,734	50	Crainary
UBS securities Australia Ltd	Buy	2,82/	709	Ordinary
JBS Securities Australia Ltd	Buy	53,134	5,489	Ordinary
S Securities Australia Ltd	Buy	19,544	2,019	Ordinary
UBS Securities Australia Ltd	Buv	1,210	125	Ordinary
C Sequrities Australia 11d	Bis	13 900	1 436	Ordinary
TIBE Consider Australia Ltd	, and	35 953	7 691	Ordinary
S Securities Australia Ltd	gng	266,62	2,001	Oldinary
is securities Australia Ltd	Buy	7,492	1/4	Ordinary
JBS Securities Australia Ltd	Buy	6,776	200	Ordinary
Securities Australia Ltd	Buy	339	35	Ordinary
URS Securities Australia Ltd	N. B	4.840	200	Ordinary
DC Contribute Australia 140	No.	A08 0	1 013	Ordinan
Dog Securities Australia Etc	S C C	000%	250.	o deligi
S Securities Australia Ltd	ans	527'/#	6/0/4	Oldinary
UBS Securities Australia Ltd	Buy	5,314	549	Ordinary
S Securities Australia Ltd	Buy	4,840	200	Ordinary
UBS Securities Australia Ltd	Buv	20.793	2.148	Ordinary
S Societies Australia I to	Bin	68 592	7.086	Ordinan
THE REPORT OF THE PARTY OF THE	600	20000	200	100
S Securities Australia Ltd	and a	n/pio	OI.	Oldinal y
JBS Securities Australia Ltd	Buy	3,8/2	400	Ordinary
JBS Securities Australia Ltd	Buy	21,354	2,206	Ordinary
UBS Securities Australia Ltd	Buy	1,859	192	Ordinary
JBS Securities Australia Ltd	Buy	3,707	383	Ordinary
BS Securities Australia Itd	Buv	7.212	745	Ordinary
O Countries Australia 140	D. S.	1 255	0.1	ording.
ODD Deculities Australia Ltd	and a	CCC'1	2 5	S C C C C C C C C C C C C C C C C C C C
JBS Securities Australia Ltd	Buy	5,663	585	Ordinary
UBS Securities Australia Ltd	Buy	28,111	2,904	Ordinary
JBS Securities Australia Ltd	Buv	44,634	4,611	Ordinary
C Constitute Australia 11d	Busy	826	101	Ordinary
o peculities Australia Liu	, and	0/6	5	100
COC Cocymition Assets in 14d				, , , , ,
1. In the second	Buy	4.124	426	Ordinary

	Late of the Assessment Control of the		1	- X-X	Cranicary
4-Jul-14	I IBS Societies Australia Ltd	E .	98 186	(5,0,0)	Ordinan
04-lul-14	LIBS Securities Australia Itd	S	2 034	(208)	Ordinary
04-1-1-1-A	I IDC Congrition American 142	= 3	10,550	(1,000)	Codings
4-101-14	Upo Securities Australia Liu	120	Occ,e	(886,1)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	3,100	(317)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	5,790	(265)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	5,790	(265)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	20.	(2)	Ordinary
04-lul-14	UBS Securities Australia I to	Sell	3 452	(353)	Ordinary
4-11-14	I IBS Securities Australia 11d	3	797.05	(202)	Ordinary
04-1-11-14	I IRS Securities Australia I td	100	975 55	(8/2 0)	Ordinary
4-1-1-4	1 IBS Societies Australia 14d	Coll	000	(2)	Crainaly
1	ODS Securities Australia LIG) Seil	07	(2)	Orginary
04-Jul-14	UBS Securities Australia Ltd	Sell	70	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Setl	20	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	20	(2)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seil	20	(2)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	9.643	(986)	Ordinary
04-1-1-14	I IRS Sequenties Australia I to	Cell	9.170	(222)	ordina.
04-1-1-14	I IRS Countries Australia 14d	Soll	170	(337)	Ordinary Ordinary
04 1.1 14	1 IDC Countition Australia 1nd		100.00		Ordinary Ordinary
11-10-1-1	Opp Securities Australia Liu	lac l	/60°06		Ciginaly
#-Inc-#	Obs Securities Australia Liu	lak Lak	115/1	(0//1)	Ordinary
U4-JUI-14	UBS Securities Australia Ltd	Sell	/63		Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	3,159)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	205		Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	1,985	(203)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	3,912	(400)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	75 956	(2,654)	Ordinary
4-11/1-14	I JRS Securities Australia Ltd	9	2 142	(510)	Ordinary
04-1-11-14	I IRS Sequirities Australia Ltd	= 9	N2C	(90)	Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Control Control Australia Australia	100	000	(103)	Oldinary Oldinary
4-JUI-14	Obs securities Austrana Ltd	Sell	876,1	(/61)	Ordinary
04-Jul-14	Securities Austrai	Sell	616	(63)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	36,646	(3,747)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	1,623	(166)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	753		Ordinary
04-1-1-14	LIRS Securities Australia Ltd	Sell	929		Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Distriction Australia Ltd	Soll	030		i i i
1 1 1 4	Late Countries Australia Liu	100	507'0		Cidillary
11.100-1	Coo Securities Australia Etc	301	701,1		Cidinaly
4-Jul-14	UBS Securities Australia Ltd	Seil	10,592	5	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Setl	3,804	(386)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	21,281	(2,176)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	841	(98)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	7,873	(802)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	11.003	(1.125)	Ordinary
1.1.1.1	I IDC Conucition Australia 14d		2127	(527)	Ordinan
1 1 1 1	Tipo complete America Fed	11-2	10.10	(25)	Couriery
04-Jul-14	ODS Securities Australia Ltd	- S	0000	(770)	Crainary
04-JUI-14	UBS Securities Australia Ltd	Sell	43,276	(7,38U)	Ordinary
74-Jul-14	UBS Securities Australia Ltd	Sell	333	(34)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	19,003	(1,943)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	12,714	(1,300)	Ordinary
24-Jul-14	UBS Securities Australia Ltd	Sell	886	(101)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	lleS	10.758	(1,100)	Ordinary
M-101-14	I IBC Cocurities Australia I to	ll 95	202	(55)	Ordinan
04-14-14	Life Countries August 19	100	2,103	(310)	Ording
N 197	bet citation Australia 140		000	(6,1)	Control of
1 1 1 1	Late of the state	===		(2)	(in in in
11-101-1	Coo securities Australia Lid	nen o	67	(c)	Cidinaly
04-JUI-14	Ubs securities Australia Ltd	Sell	4,130	(474)	Orginary
04-Jul-14	UBS Securities Australia Ltd	9			
****		1150	460	(47)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	460	(2)	Ordinary

40,937 Ordinary		1,202 Ordinary		150,329 Ordinary	L		L	L	L	(70) Ordinary		(7.473) Ordinary			(1) Ordinal		(1 705) Ordinary	L			(400) Ordinary			(137) Ordinary						(54) Ordinary			(912) Ordinary			1		(4/1) Ordinary				(701) Ordinary		(9,648) Ordinary			(3.007) Ordinary	L		L		(420) Ordinary	(5) Ordinary					(300) Ordinary (300) Ordinary
400,257	10,758	11,/44	876,657	1,472,548	183,431	9.412	36.397	4.302	97 971	989	2.450	73 235	55.066	20,000	01	2 8	17 591	3 165	2,744	5,106	3,920	118	9//07	ε7 L	6,027	49	4,920	109,045	34,672	675	0,958	2.715	8,938	5,419	39,677	1,878	157,096	4,609	4.293	20	31,717	6,856	6,944	94,35/	6474	6.484	29,408	15,315	6,944	3,912	199,502	4,108	49	3.159	39		3,912	3,912 2,934
8uy	Buy	Buy	Je Se	Buy	Buy	A S	Buv	Birk	Jan S	les.	Sel	Cell	3	lio o	Sall	Sell	ilec des	i i	Set	Sell	Self	Sell	i Ke	la Z	Sell	Sell	Sell	Sell	Sell	les.	Sell Call	Set	Sell	Self	Sell	Sell	Sell	Sell Sell	Sell	Sell	Sell	Sell	Seli	Seli	Zel!	iğ d	Sell	Seil	Sett	Self	Sell	Sell	Sell	ije.	Sell		- S	Sell
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Wealth Management Australia Ltd	UBS Wealth Management Australia Ltd	I IBS Sequities Australia I td	I IBS Societies Australia I to	I BS Sequirities Australia I to	DO Constitution Americal 144	I IBS Societies Australia I td	I IBS Securities Australia 14d	1 IDC Countries Australia Ltd	UDS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	1 IRS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IRS Securities Australia Ltd	IUBS Securities Australia Ltd	In the Contract Assessment of the	UBS Securities Australia Liu	UBS Securities Australia Ltd UBS Securities Australia Ltd
04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-14-14	04-10-14	ı		77.77	П	ı	04-101-14	ı	04-JUI-14	1	04-101-14			04-Jul-14	- 1	04-Jul-14	04-501-14	04-3ul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14		1	04-1st-14	ı		04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-111-14	04-101-14	04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-101-14	04-301-14	04-101-14	04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-Jul-14	04-101-14	04-101-14		04-Jul-14	04-Jul-14 04-Jul-14

1.4.4	The Committee Account to		-		Circinary
04-lul-14	LIBS Securities Australia Itd	100	of.	(9)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	381	(5)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	lleS.	7.677	(785)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	5 809	(1594)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	205	(21)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	90,778	(9,282)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	2,934	(300)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Self	5,359	(548)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Self	587	(09)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	2,142	(219)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	999	(89)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	17,809	(1.821)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Self	6,944	(710)	Ordinary
04-Jui-14	UBS Securities Australia Ltd	Seli	2,729	(279)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	-Se	5,359	(548)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	3.912	(400)	Ordinary
-Jul-14	UBS Securities Australia Ltd	les.	16.626	(1,700)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	14.973	(1531)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	2,729	(279)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	4.098	(419)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	10.269	(1 050)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	1 359	(139)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	20.245	(07.070)	Ordinary
-Jul-14	UBS Securities Australia Itd	les	1 751	(179)	Ordinary
04-341-14	UBS Securities Australia Itd	Sell	1.056	(801)	Ordinary
34-Jul-14	UBS Securities Australia 11d	Sell	2,171	(222)	Ordinary
-Jul-14	UBS Securities Australia Itd	lias	645	(66)	Ordina
04-101-14	IJRS Securities Australia 11d	3	5,634	(575)	Ordinary
111-14	LIRS Securities Australia 14d	199	3,042	(115)	Ordinary
111-14	I IRS Securities Australia 14d	Coll	250,0	(37.0.0)	Ordinary
1	A IDC Committee Australia Ltd	1100	162,62	(2,370)	Gustary
04-1-1-14	Description Application of the	100	3,032	(166)	Ordinary
1111	Tips Counting Australia Liu	201	24.30	(67)	Gainary
1 1	Los comities Australia Ltd	llac llac	521,62	(806,2)	Orginary
04-1-11	TOS Comitino Australia Lia	Jes Jes	012,2	(977)	Ordinary
1 100	COS SECULICIES AUSTRAINA LIC	196	2,010	(200)	Ordinary
JUI- 14	UBS Securities Australia Ltd	Seli	2,015	(506)	Ordinary
24-Jul-14	UBS Securities Australia Ltd	Sell	866	(102)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	1,897	(194)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seli	3,159	(323)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	21,907	(2,240)	Ordinary
-Jui-14	UBS Securities Australia Ltd	Sell	3,149	(322)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	3,149	(322)	Ordinary
Jul-14	UBS Securities Australia Ltd	lleS	1,897	(194)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	870	(88)	Ordinary
)4-11-14	HBS Securities Australia Ltd	a de la companya de l	2 064	(117)	, de cipao
04-101-14	LIBS Securities Australia 14d	llo2	2,004	(300)	Gurary
11,11	LIBS Securities Australia 14d	Col	2 012	(007)	Ordinary Ordinary
04-14-14	LIBS Securities Australia (14)	3	21.00	(54)	Ordinary
11.17	I IBC Socurities Australia 14d	Tool O	3 150	(600)	ording.
24-3-1-14	I IBS Securities Australia 11d	100	147	(15)	Californ
04-hid-14	I BS Secretion Australia 11d	le o	190	(5.1)	ording)
1,1,1	Library Australia	100	2017	(00)	Quilaiy
	COO Securities Australia Lio		1, 00	(420)	oldinal y
04-101-14	Ups securities Australia Ltd	liac l	1,731	(//)	Ordinary
04-101-14	UBS Securities Australia Ltd	Nell Nell	48,900	(2,000)	Ordinary
JUI-14	UBS Securities Australia Ltd	Sell	28,157	(2,879)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	929	(65)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sel	1,644	(168)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seli	979	(64)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	3,239	(331)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	832	(82)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seli	14,678	(1,500)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Self	082'6	(1,000)	Ordinary
24-Jul-14	UBS Securities Australia Ltd	Seli	162.0	(100)	,acrip-C
			7//2	(100)	CELIE
04-Jul-14	UBS Securities Australia Itd	ll d'S	27,2	(235)	Ordinary

04-Jul-14	Let Constitute Assessment 144	lion	9		
04-11-14	UBS Securities Australia 11d	F 19	186	(19)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	-Se-	10.318	(1.055)	Ordinary
04-1ul-14	I IRS Securities Australia ! td		780	(00)	Ordinan
11.10	ODS Securities Australia Etu	368	407	(67)	Gorrary
-1-101-1-1-0	Obs securities Australia Liu	ii o	705	(/6)	Orginary
-101-114	Obs securities Australia Ltd	Sell	4,450	(455)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	6/8'1	(141)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	1,897	(194)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	3,834	(365)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	9,780	(1,000)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	95	082.6	(1,000)	Ordinary
04-111-14	UBS Securities Australia Itd	وا	298	(37)	Ordinary
1	I IDC Contrition Assettation 14d	200	4 460	(457)	, in the contract of
	Let offered Assessed 190	100	010	(101)	Cidillary
#1-Inf	Obs Securities Australia Ltd	Jen Jen Jen Jen Jen Jen Jen Jen Jen Jen	516	(AP)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seil	7,697	(787)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	323	(33)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	170,95	(366'E)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	1,555	(129)	Ordinary
-h1-14	UBS Securities Australia Ltd	Sell	4.626	(473)	Ordinary
V 1-4	1 IBS Securities Australia I to	<u></u>	1 201	(132)	Ordinan
	1 DC Coomition Australia 14d	1100	03	(2)	Tall and
04-701-14	מפס שבתו ווגבא שתאו מוום דנת	Je	00 4	(/)	Cidinaly
JUI-14	UBS Securities Australia Ltd	Sell	196,8	(405)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	6,113	(625)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	9/6/6	(1,020)	Ordinary
Jul-14	UBS Securities Australia Ltd	-Sell	10,113	(1,034)	Ordinary
Jul-14	UBS Securities Australia Ltd	=95	2003	(521)	Ordinary
04-8-1-14	1 BS Securities Australia Ltd	llay	4117	(1771)	Ordinary
0.4 6.1.1.4	i IDC Coourition Australia 14d	Coll	0	(10)	Ordinan
1 1 1 1	Tipo Commission Australia 144	===	1 252	(90.1)	A STATE OF
1	ODS Securities Australia Ltd	- N	767'1	(971)	Ordinary
JUI-14	UBS Securities Australia Ltd	Xe.	817,11	(1,147)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	886	(101)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sett	7,110	(727)	Ordinary
04-lul-14	UBS Securities Australia Ltd	lla ₂	593	(UE)	Ordinary
	100 Car mistar Australia I tal	1)00	000	(200)	
+1-Inc-to	COS Seculifies Australia Ltd	Dell Dell	004,0	(000)	Clossaly
1-10	Obs Securities Australia Lid	llac llac	0,510	(040)	Orginary
04-JUI-14	UBS Securities Australia Ltd	Sell	156	(16)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	1,076	(110)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	6,122	(929)	Ordinary
04-101-14	UBS Securities Australia Ltd	Sell	866.9	(715)	Ordinary
04-1-1-14	LIRS Securities Australia I to	97	3 159	(323)	Ordinany
	Day of the American Cook		CELO JC	(020 0)	1000
7U-14	UBS Securities Australia Ltd	Sell	5/8/5	(3,558)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	9,897	(1,012)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seli	068	(16)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	44.988	(4.600)	Ordinary
11.1	LIBS Societies Australia I to	Coll	121	(13)	Ordinary
	Late of the state		(7)	(00)	Calling y
04-JUI-14	UBS Securities Australia Ltd	Sell	793	(30)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	978	(100)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	215	(22)	Ordinary
24-Jul-14	UBS Securities Australia Ltd	Sell	303	(31)	Ordinary
04-10-14	UBS Securities Australia I td	روا	8 724	(892)	Ordinary
11.1.1	I IBC Cocurities Australia 14d	lo S	750	(40)	Coding
	COO Secondary Assessment Land	1.0	100	(101)	7
04-101-14	UBS Securities Australia Ltd) Sell	1,02,1	(cn)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	5,115	(523)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	7,433	(200)	Ordinary
04-10-14	LIBS Securities Australia Itd	llas	5.751	(588)	Ordinary
04-1-1-14	I IBS Societities Australia I to	lla)	30 844	(1/107/1	Ordinan
	COD OCCUPATION A CONTROL AND A		10,00	(1,0,1)	Oldinary Oldinary
04-JUI-14	UBS Securities Australia Ltd	Sell	5,584	(1/4)	Ordinary
Jul-14	UBS Securities Australia Ltd	Sell	36,812	(3,764)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	3,863	(362)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seil	8.039	(822)	Ordinary
24-3-11-14	I IRS Securities Australia I to	ije)	9.017	(000)	Ordinary
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1	245	(101)	in in it
JUI-14	UBS Securities Australia Ltd	Seri	217,5	(284)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seil	4,264	(436)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	9.271	(0/0)	Cadinan
1.1.4.4				(010)	Crainary
	It for the mitigal Attention 1+4	lico.	000	(046)	Ordings
04-JUI-14	UBS Securities Australia Ltd	Sell	489	(50)	Ordinary

27 77	Lat Late A consistence of the Late Late Late Late Late Late Late Lat	1-0	CEC	(66)	5 0
Ι.	1.85 Securities Australia Ltd	Sel.	1 771	(371)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	2,396	(245)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	6,386	(653)	Ordinary
D4-Jul-14	UBS Securities Australia Ltd	Self	7,980	(816)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Self	137	(14)	Ordinary
L	UBS Securities Australia Ltd	lleS.	998'9	(702)	Ordinary
04-3ul-14	UBS Securities Australia Ltd	ll-8S	2,142	(219)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	lieS.	21,369	(2.185)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	les!	826	(86)	Ordinary
)4-}ul-14	UBS Securities Australia Ltd	Sell	11,736	(1.200)	Ordinary
4-Jui-14	UBS Securities Australia Ltd	Sell	6,249	(639)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	lles	802	(82)	Ordinary
L	UBS Securities Australia Ltd	Sell	8,430	(862)	Ordinary
4-1-14	UBS Securities Australia Ltd	Sell	12,616	(1,290)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	7,824	(800)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	6,807	(969)	Ordinary
J4-Jul-14	UBS Securities Australia Ltd	Sell	12,499	(1,278)	Ordinary
	UBS Securities Australia Ltd	Sell	8,802	(006)	Ordinary
74-Jul-14	UBS Securities Australia Ltd	Sell	323	(33)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	986,9	(653)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	4,323	(442)	Ordinary
)4-Jul-14	UBS Securities Australia Ltd	Sell	10,846	(1,109)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	274	(28)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	826	(86)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Sell	1,457	(149)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	10	Û	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Setl	333	(34)	Ordinary
)4-Jul-14	UBS Securities Australia Ltd	Sell	1.848	(189)	Ordinary
L	UBS Securities Australia Ltd	Seil	1.046	(107)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Seil	264	(72)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	23.619	(2.415)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Sell	6,924	(208)	Ordinary
14-Jul-14	UBS Securities Australia Ltd	Sell	8,792	(668)	Ordinary
J4-Jul-14	UBS Securities Australia Ltd	Sell	3,276	(335)	Ordinary
14-Jul-14	UBS Securities Australia Ltd	les.	812	(83)	Ordinary
14-Jul-14	UBS Securities Australia Ltd	Sell	29	(3)	Ordinary
34-Jul-14	UBS Securities Australia Ltd	Seil	2,103	(215)	Ordinary
14-Jul-14	UBS Securities Australia Ltd	Seil	2.142	(219)	Ordinary
4-Jul-14	UBS Securities Australia Ltd	Seil	989	(99)	Ordinary
14-Jul-14	UBS Securities Australia Ltd	Şeil	3.912	(400)	Ordinary
L	UBS Securities Australia Ltd	Sell	2.582	(264)	Ordinary
24-Jul-14	UBS Securities Australia Ltd	Sell	1,800	(184)	Ordinary
L	UBS Securities Australia Ltd	Sell	19,853	(2,030)	Ordinary
4-311-14	UBS Securities Australia Ltd	Sell	866	(102)	Ordinary
04-1-14 04-1-14	1 IRS Sequirities Australia 1 to	::51 	8 333	(852)	Ording.
14-feel-14	I IRC Securities Australia Ltd	Soil	22 711	(3.245)	Ordinary
١,	J IRC Constitute Australia 14d	Coil	22,714	(3,243)	Grainary
04-101-14 04-14-14	UBS Securities Australia Ltd	liac	1 056	(100)	Ordinary
04-101-14 04-101-14	1 IBC Constition Australia 14d	lisc lisc	000'-	(202)	Ordinary
04-101-12	1 IDC Countition Australia Ltd	1100	0,100	(575)	Ordinary
04-101-14 04-14-14	I BC Contribute Australia 144	lisc	2,133	(525)	Ordinary
04-1-1-1	LIBC Counties Australia Ltd	lisc	2,133	(10)	Ordinary
١,	Libe Counties Australia Ltd	lier lier	00.00	(01)	Garriary
04-101-14	Use comiting Australia Ltd	IIAC IIAC	5,139	(360)	Ordinary
04-101-14	1 DC Cocurition Australia 144	lisc	175'	(507)	Ordinary
04-1-1-1	Libe countries Australia Ltd	les.	11 765	(1 203)	Ordinary
1 1 1 1	Libe constitute Australia Ltd	les.	7120	(1,00)	Ordinary
04-1-1-14	I BC Counties Australia 14d	100	200	(10)	Ordinary
١.	Libe Counties Australia Ltd	nac live	1000	(6)	Oldinary
04-101-14	USS Securities Australia Ltd	nes.	0,924	(202)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	17/	(13)	Ordinary
_ [UBS Securities Australia Ltd	Sell	773	(79)	Ordinary
	UBS Securities Australia Ltd	Sell	225	(23)	Ordinary
04-Jul-14	UBS Securities Australia Ltd	Sell	4.127	(CCF)	Ordinany
I.				(77+)	5
04-14-14	II IBS Securities Australia Ltd	Call	303	(31)	Ordinary

Sell Sell	
Sell	
Ē	
<u></u>	
195	
Seil	
Sell	
Selt	
Sell	
- Rei	
130	l
les.	
38	
Sell	
Sell	
Sell	
les.	
- S	
les o	
ķ 3	
1	
وا	
Seil	
<u>se</u>	
Sell	
≅	
Sell	
Sell	
Self	
Self	
## S	
100	
Sell	
Self	
Sett	
Zell Sell	
Jet J	
Neil Sell	
R 3	
3	
<u></u>	
3	
Sell	
ll-S.	
BS	
Sell	
Sell	
les	
Sell	
Sell	
Sell	
les	
Sell	
3	
130	
100	
lac lac	
× 3	
183	
IIAC I	

USS Securities Australia LU SM 9,280 (1,000) USS Securities Australia LU SMI 2,834 (3,702) (7,13) USS Securities Australia LU SMI 2,834 (3,702) (7,13) USS Securities Australia LU SMI 2,834 (3,202) (7,13) USS Securities Australia LU SMI 2,834 (3,202) (7,13) USS Securities Australia LU SMI 2,811 4,205 (4,202) USS Securities Australia LU SMI SMI 4,205 (4,202) USS Securities Australia LU SMI SMI 6,210 (4,202) USS Securities Australia LU SMI SMI 6,210 (4,202) USS Securities Australia LU SMI SMI 1,271 (4,202) USS Securities Australia LU SMI SMI 1,271 (4,602) USS Securities Australia LU SMI SMI 1,271 (4,602) USS Securities Australia LU SMI SMI 1,271 (4,602) USS Securities Australia LU </th <th>74- fr il -14</th> <th>ILIBS Sacrimitiae Australia Ltd</th> <th></th> <th></th> <th></th> <th>/UKCIDAL</th>	74- fr il -14	ILIBS Sacrimitiae Australia Ltd				/UKCIDAL
USS Securities Australia Ltd Sell	1-Jul-14	UBS Securities Australia 11d	ig y	087.0	(0001)	Coling of
USS Securities Australia Ltd Sell	1-Jul-14	UBS Securities Australia Ltd	ll de	6 973	(713)	Ordinary
USS Securites Australia Ltd Sell	1-Jul-14	UBS Securities Australia 14d	9	V60 C	(010)	Ordinary
USS Scontries Australia LID Sell 4,205 (430) USS Scontries Australia LID Sell 4,205 (430) USS Scontries Australia LID Sell 6,210 (450) USS Scontries Australia LID Sell 6,210 (5,200) USS Scontries Australia LID Sell 6,210 (5,200) USS Scontries Australia LID Sell 6,210 (5,200) USS Scontries Australia LID Sell 1,223 (6,520) USS Scontries Australia LID Sell 1,233 (6,520) USS Scontries Australia LID Sell 1,523 (6,520) USS Scontries Australia LID Sell 1,523 (6,520) USS Scontries Australia LID Sell 1,524 (7,71) USS Scontries Australia LID Sell 1,524 (7,620) USS Scontries Austra	- hil-14	BS Societies Australia 14d	100	2,334	(000)	Ordinary
655 Scorrifes Australia Ltd Sell 4,489 (459) UBS Scorrifes Australia Ltd Sell 6,518 (459) UBS Scorrifes Australia Ltd Sell 6,518 (5,500) UBS Scorrifes Australia Ltd Sell 6,510 (6,500) UBS Scorrifes Australia Ltd Sell 6,510 (7,710) UBS Scorrifes Australia Ltd Sell 6,520 (7,620) UBS Scorrifes Australia Ltd Sell 7,521 (7,620) UBS Scorrifes Aus	Lhil-14	I IR Securities Australia 14d	lis Soll	300 4	(766)	Oronary
USS Securities Australia Ltd Sell Sell Sel S	1-Aut-14	UBS Securities Australia Ltd	900	000 /	(004)	Ordinary
UBS Securities Australia Ltd Sell 66,233 6,528 UBS Securities Australia Ltd Sell 6,589 6,589 UBS Securities Australia Ltd Sell 5,890 1,652 UBS Securities Australia Ltd Sell 5,891 1,527 529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,529 1,529 1,529 UBS Securities Australia Ltd 5,891 5,891 1,591 1,	-Jul-14	UBS Securities Australia Ltd	Sell S	OSE U	(500)	Ordinary
UBS Securities Australia Ltd Seil 6.210 (6.52) UBS Securities Australia Ltd Seil 6.210 (6.50) UBS Securities Australia Ltd Seil 15.240 (7.625) UBS Securities Australia Ltd Seil 15.242 (7.625) UBS Securities Australia Ltd Seil 15.242 (7.626) UBS Securities Australia Ltd Seil 15.242 (7.626) UBS Securities Australia Ltd Seil 15.242 (7.626) UBS Securities Australia Ltd Seil 15.242 (4.626) UBS Securities Australia Ltd Seil 15.242 (4.626) UBS Securities Australia Ltd Seil 15.252 (4.626) UBS Securities Australia Ltd Seil 15.253 (4.626) UBS Securities Australia Ltd Seil 15.254 (4.626) UBS Securities Australia Ltd Seil 15.254 (4.626) UBS Securities Australia Ltd Seil 15.274 (1.627) UBS Securities Australia Ltd Seil 15.274 (1.627)	-lul-14	I BS Securities Australia Ltd	la Coll	555.62	(046)	OCCURIENCE
UBS Securities Australia Ltd Sell 16,250 (6,500) UBS Securities Australia Ltd Sell 15,314 (1,625) UBS Securities Australia Ltd Sell 12,714 (1,635) UBS Securities Australia Ltd Sell 3,539 (1,625) UBS Securities Australia Ltd Sell 3,539 (1,625) UBS Securities Australia Ltd Sell 3,539 (1,626) UBS Securities Australia Ltd Sell 3,540 (5,68) UBS Securities Australia Ltd Sell 3,540 (6,69) UBS Securities Australia Ltd Sell 3,540 (6,67) UBS Securities Australia Ltd Sell 1,531 (1,67) UBS Securities Australia Ltd Sell 1,542 (1,67) UBS Securities Australia Ltd Sell 1,542 (1,67) UBS	1-Jul-14	UBS Securities Australia Ltd	100	6.210	(0/5/0)	Critical
URS Securities Australia Ltd Sell (1,623) (1,623) URS Securities Australia Ltd Sell 1,539 (1,622) URS Securities Australia Ltd Sell 1,539 (1,622) URS Securities Australia Ltd Sell 5,242 (4,626) URS Securities Australia Ltd Sell 5,540 (4,626) URS Securities Australia Ltd Sell 5,540 (4,626) URS Securities Australia Ltd Sell 5,540 (4,627) URS Securities Australia Ltd Sell 3,540 (4,627) URS Securities Australia Ltd Sell 3,540 (4,637) URS Securities Australia Ltd Sell 3,540 (4,637) URS Securities Australia Ltd Sell 3,540 (4,637) URS Securities Australia Ltd Sell 1,627 (4,637) URS Securities Australia Ltd Sell 1,627 (4,637) URS Securities Australia Ltd Sell 3,640 (4,537) URS Securities Australia Ltd Sell 3,640 (4,537)	-Jul-14	UBS Securities Australia 11d	19	000 87	(000)	Carling C
UBS Securities Australia Ltd Sell 1.6.210 (1.6.2) UBS Securities Australia Ltd Sell 5.359 (6.68) UBS Securities Australia Ltd Sell 5.359 (6.68) UBS Securities Australia Ltd Sell 5.359 (6.68) UBS Securities Australia Ltd Sell 5.50 (7.62) UBS Securities Australia Ltd Sell 5.50 (7.62) UBS Securities Australia Ltd Sell 5.50 (7.62) UBS Securities Australia Ltd Sell 3.53 (6.69) UBS Securities Australia Ltd Sell 3.53 (6.69) UBS Securities Australia Ltd Sell 3.59 (6.69) UBS Securities Australia Ltd Sell 3.69 (6.69) UBS Securities Austra	1-Jul-14	UBS Securities Australia 11d	9	15 803	(1625)	or dings
UBS Securities Australia Ltd Sell 1,2714 (1,350) UBS Securities Australia Ltd Sell 5,322 (6,48) UBS Securities Australia Ltd Sell 6,372 (6,68) UBS Securities Australia Ltd Sell 6,372 (7,68) UBS Securities Australia Ltd Sell 1,5824 (1,618) UBS Securities Australia Ltd Sell 1,5824 (1,618) UBS Securities Australia Ltd Sell 1,5824 (1,618) UBS Securities Australia Ltd Sell 1,582 (1,618) UBS Securities Australia Ltd Sell 1,582 (1,618) UBS Securities Australia Ltd Sell 1,582 (1,68) UBS Securities Australia Ltd Sell 1,593 (1,68) UBS Securities Australia Ltd Sell 1,593 (1,68) UBS Securities Australia Ltd Sell 1,593 (1,69) UBS Securities Australia Ltd Sell 2,455 (1,68) UBS Securities Australia Ltd Sell 2,545 (1,68) UBS	1-Jul-14	UBS Securities Australia Ltd	Sell	6.210	(635)	Ordinary
UBS Securities Australia Ltd Sell 5,324 (4,256) UBS Securities Australia Ltd Sell 6,324 (4,626) UBS Securities Australia Ltd Sell 20 (7) UBS Securities Australia Ltd Sell 156 (7) UBS Securities Australia Ltd Sell 3,540 (4626) UBS Securities Australia Ltd Sell 3,540 (4627) UBS Securities Australia Ltd Sell 3,540 (4628) UBS Securities Australia Ltd Sell 3,540 (4629) UBS Securities Australia Ltd Sell 3,540 (4629) UBS Securities Australia Ltd Sell 3,540 (4639) UBS Securities Austral	-Jul-14	UBS Securities Australia Ltd	Sell	12.714	(1 300)	Ordinary
UBS Securities Australia Ltd Sell 4,524 (4,526) UBS Securities Australia Ltd Sell 6,24 (4,526) UBS Securities Australia Ltd Sell 6,78 (7,10) UBS Securities Australia Ltd Sell 1,582.4 (1,61) UBS Securities Australia Ltd Sell 1,582.7 (4,69) UBS Securities Australia Ltd Sell 1,521 (1,503) UBS Securities Australia Ltd Sell 1,521 (1,503) UBS Securities Australia Ltd Sell 1,522 (1,693) UBS Securities Australia Ltd Sell 3,162 (1,693)	-Jul-14	UBS Securities Australia Ltd	Sell	3,834	(392)	Ordinary
UBS Securities Australia Ltd Sell 45,242 (4,626) UBS Securities Australia Ltd Sell 20 (2) UBS Securities Australia Ltd Sell 156 (7) UBS Securities Australia Ltd Sell 3,540 (56) UBS Securities Australia Ltd Sell 3,540 (56) UBS Securities Australia Ltd Sell 3,540 (57) UBS Securities Australia Ltd Sell 3,550 (46) UBS Securities Australia Ltd Sell 3,550 (46) UBS Securities Australia Ltd Sell 3,550 (46) UBS Securities Australia Ltd Sell 1,521 (1,503) UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Ltd	Sell	5,359	(548)	Ordinary
UBS Securities Australia Ltd Sell 70 (7) UBS Securities Australia Ltd Sell 15,824 (1,618) UBS Securities Australia Ltd Sell 15,824 (1,618) UBS Securities Australia Ltd Sell 3,335 (1,618) UBS Securities Australia Ltd Sell 3,335 (4,691) UBS Securities Australia Ltd Sell 2,545 (4,691) UBS Securities Australia Ltd Sell 1,547 (4,691) UBS Securities Australia Ltd Sell 1,547 (4,691) UBS Securities Australia Ltd Sell 1,547 (4,692) UBS Securities Australia Ltd Sell 1,547 (4,692) UBS Securities Australia Ltd Sell 1,672 (4,692) UBS Securities Australia Ltd Sell 1,672 (4,692) UBS Securities Australia Ltd Sell 3,162 (5,892) UBS Securities Australia Ltd Sell 3,162 (5,892) UBS Securities Australia Ltd Sell 3,162 (5,892) UBS	-Jul-14	UBS Securities Australia Ltd	Sell	45,242	(4,626)	Ordinary
UBS Securities Australia Ltd Sell 158.24 (1,618) UBS Securities Australia Ltd Sell 1,582.4 (1,618) UBS Securities Australia Ltd Sell 3,540 (652) UBS Securities Australia Ltd Sell 3,540 (652) UBS Securities Australia Ltd Sell 1,291 (1,618) UBS Securities Australia Ltd Sell 1,291 (1,631) UBS Securities Australia Ltd Sell 1,291 (1,632) UBS Securities Australia Ltd Sell 1,291 (1,632) UBS Securities Australia Ltd Sell 1,291 (1,637) UBS Securities Australia Ltd Sell 1,630 (4,630) UBS Securities Australia Ltd Sell 1,630 (4,630) UBS Securities Australia Ltd Sell 1,631 (4,630) UBS Securities Australia Ltd Sell 1,630 (4,630) UBS Securities Australia Ltd Sell 1,630 (4,630) UBS Securities Australia Ltd Sell 1,631 (4,630) <td< td=""><td>-Jul-14</td><td>UBS Securities Australia Ltd</td><td>les</td><td>20</td><td>(2)</td><td>Ordinary</td></td<>	-Jul-14	UBS Securities Australia Ltd	les	20	(2)	Ordinary
UBS Securities Australia Ltd Sell 156 (16) UBS Securities Australia Ltd Sell 3,440 (162) UBS Securities Australia Ltd Sell 3,440 (162) UBS Securities Australia Ltd Sell 6,787 (691) UBS Securities Australia Ltd Sell 7,257 (691) UBS Securities Australia Ltd Sell 1,291 (158) UBS Securities Australia Ltd Sell 1,291 (150) UBS Securities Australia Ltd Sell 1,292 (143) UBS Securities Australia Ltd Sell 1,293 (143) UBS Securities Australia Ltd Sell 1,294 (1,210) UBS Securities Australia Ltd Sell 1,294 (1,210) UBS Securities Australia Ltd Sell 2,545 (5,50) UBS Securities Australia L	-Jul-14	UBS Securities Australia Ltd	Sell	89	(2)	Ordinary
UBS Securities Australia Ltd Sell 15.824 (1618) UBS Securities Australia Ltd Sell 5.40 (32) UBS Securities Australia Ltd Sell 6.756 (691) UBS Securities Australia Ltd Sell 4.58 (691) UBS Securities Australia Ltd Sell 3.35 (341) UBS Securities Australia Ltd Sell 3.52 (368) UBS Securities Australia Ltd Sell 2.545 (168) UBS Securities Australia Ltd Sell 2.545 (168) UBS Securities Australia Ltd Sell 1.009 (1.057) UBS Securities Australia Ltd Sell 1.009 (1.057) UBS Securities Australia Ltd Sell 4.74 (1.057) UBS Securities Australia Ltd Sell 1.009 (1.057) UBS Securities Australia Ltd Sell 1.009 (1.057) UBS Securities Australia Ltd Sell 2.7291 (5.800) UBS Securities Australia Ltd Sell 3.162 (5.800) UBS Securities Au	-Jul-14	UBS Securities Australia Ltd	Sell	951	(16)	Ordinary
UBS Securities Australia Ltd Sell 3540 (82) UBS Securities Australia Ltd Sell 6,788 (691) UBS Securities Australia Ltd Sell 6,788 (691) UBS Securities Australia Ltd Sell 1,563 (168) UBS Securities Australia Ltd Sell 1,231 (168) UBS Securities Australia Ltd Sell 1,231 (46) UBS Securities Australia Ltd Sell 1,231 (46) UBS Securities Australia Ltd Sell 1,231 (45) UBS Securities Australia Ltd Sell 1,039 (45) UBS Securities Australia Ltd Sell 4,210 (45) UBS Securities Australia Ltd Sell 4,210 (45) UBS Securities Australia Ltd Sell 4,210 (45) UBS Securities Australia Ltd Sell 4,341 (45) UBS Securities Australia Ltd Sell 5,365 (5,29) UBS Securities Australia Ltd Sell 5,365 (5,57) UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Ltd	Sell	15,824	(1,618)	Ordinary
UBS Securities Australia Ltd Sell 6.758 (691) UBS Securities Australia Ltd Sell 6,758 (691) UBS Securities Australia Ltd Sell 4,587 (469) UBS Securities Australia Ltd Sell 1,525 (23) UBS Securities Australia Ltd Sell 1,525 (23) UBS Securities Australia Ltd Sell 1,231 (1,637) UBS Securities Australia Ltd Sell 1,231 (1,637) UBS Securities Australia Ltd Sell 1,034 (1,637) UBS Securities Australia Ltd Sell 1,034 (1,637) UBS Securities Australia Ltd Sell 1,039 (1,037) UBS Securities Australia Ltd Sell 1,039 (1,037) UBS Securities Australia Ltd Sell 1,039 (1,037) UBS Securities Australia Ltd Sell 1,034 (1,037) UBS Securities Australia Ltd Sell 3,162 (1,037) UBS Securities Australia Ltd Sell 3,162 (1,037) UBS Securi	-Jul-14	UBS Securities Australia Ltd	Sell	3,540	(362)	Ordinary
UBS Securities Australia Ltd Sell 6,78 (691) UBS Securities Australia Ltd Sell 4,587 (469) UBS Securities Australia Ltd Sell 1,231 (168) UBS Securities Australia Ltd Sell 1,231 (45) UBS Securities Australia Ltd Sell 1,231 (45) UBS Securities Australia Ltd Sell 1,231 (45) UBS Securities Australia Ltd Sell 1,4714 (1,532) UBS Securities Australia Ltd Sell 1,4714 (1,532) UBS Securities Australia Ltd Sell 1,632 (1,632) UBS Securities Australia Ltd Sell 1,632 (1,532) UBS Securities Australia Ltd Sell 1,632 (1,532) UBS Securities Australia Ltd Sell 1,632 (1,532) UBS Securities Australia Ltd Sell 3,643 (1,632) UBS Securities Australia Ltd Sell 3,643 (1,632) UBS Securities Australia Ltd Sell 3,643 (1,632) UBS Securiti	-Jul-14	UBS Securities Australia Ltd	Sell	313	(32)	Ordinary
UBS Securities Australia Ltd Sell 4.587 (469) UBS Securities Australia Ltd Sell 1.643 (168) UBS Securities Australia Ltd Sell 1.643 (168) UBS Securities Australia Ltd Sell 1.291 (132) UBS Securities Australia Ltd Sell 4.745 (46) UBS Securities Australia Ltd Sell 4.210 (1037) UBS Securities Australia Ltd Sell 4.210 (409) UBS Securities Australia Ltd Sell 4.74 (1037) UBS Securities Australia Ltd Sell 1.637 (125) UBS Securities Australia Ltd Sell 1.753 (125) UBS Securities Australia Ltd Sell 2.45 (25) UBS Securities Australia Ltd Sell 3.162 (144) UBS Securities Australia Ltd </td <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>852'9</td> <td>(1691)</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Sell	852'9	(1691)	Ordinary
UBS Securities Australia Ltd Sell 1,643 (168) UBS Securities Australia Ltd Sell 1,643 (168) UBS Securities Australia Ltd Sell 225 (36) UBS Securities Australia Ltd Sell 2,545 (250) UBS Securities Australia Ltd Sell 2,545 (250) UBS Securities Australia Ltd Sell 2,545 (250) UBS Securities Australia Ltd Sell 4,714 (1,53) UBS Securities Australia Ltd Sell 4,714 (1,53) UBS Securities Australia Ltd Sell 4,494 (459) UBS Securities Australia Ltd Sell 3,65 (1,63) UBS Securities Australia Ltd Sell 3,162 (1,63) UBS Securities Australia	-Jul-14	UBS Securities Australia Ltd	Sell	4,587	(469)	Ordinary
UBS Securities Australia Ltd Sell 1.643 (168) UBS Securities Australia Ltd Sell 352 (35) UBS Securities Australia Ltd Sell 1.291 (122) UBS Securities Australia Ltd Sell 1.271 (450) UBS Securities Australia Ltd Sell 4.210 (460) UBS Securities Australia Ltd Sell 4.210 (461) UBS Securities Australia Ltd Sell 4.210 (465) UBS Securities Australia Ltd Sell 4.249 (450) UBS Securities Australia Ltd Sell 1.622 (450) UBS Securities Australia Ltd Sell 1.782 (424) UBS Securities Australia Ltd Sell 3.162 (323) UBS Securities Australia Ltd Sell 3.162 (434) UBS Securities Australia Ltd <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>SEE'E</td> <td>(341)</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Sell	SEE'E	(341)	Ordinary
UBS Securities Australia Ltd Sell 225 (35) UBS Securities Australia Ltd Sell 1,291 (132) UBS Securities Australia Ltd Sell 1,291 (132) UBS Securities Australia Ltd Sell 4,510 (132) UBS Securities Australia Ltd Sell 1,714 (1,503) UBS Securities Australia Ltd Sell 4,434 (1,657) UBS Securities Australia Ltd Sell 4,434 (1,657) UBS Securities Australia Ltd Sell 4,434 (1,657) UBS Securities Australia Ltd Sell 5,455 (459) UBS Securities Australia Ltd Sell 5,455 (459) UBS Securities Australia Ltd Sell 5,453 (455) UBS Securities Australia Ltd Sell 5,453 (455) UBS Securities Australia Ltd Sell 5,453 (455) UBS Securities Australia Ltd Sell 5,453 (5,57) UBS Securities Australia Ltd Sell 5,453 (5,657) UBS Securities Austr	Jul-14	UBS Securities Australia Ltd	Sell	1,643	(168)	Ordinary
UBS Securities Australia Ltd Sell 35.2 (36) UBS Securities Australia Ltd Sell 1,291 (132) UBS Securities Australia Ltd Sell 450 (46) UBS Securities Australia Ltd Sell 1,4714 (1,631) UBS Securities Australia Ltd Sell 1,039 (1,632) UBS Securities Australia Ltd Sell 4,494 (459) UBS Securities Australia Ltd Sell 4,494 (459) UBS Securities Australia Ltd Sell 4,494 (459) UBS Securities Australia Ltd Sell 3,162 (38) UBS Securities Australia Ltd Sell 3,162 (35) UBS Securities Australia Ltd Sell 3,162 (35) UBS Securities Australia Ltd Sell 3,162 (35) UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Ltd	Sell	225	(23)	Ordinary
UBS Securities Australia Ltd Sell 1,291 (132) UBS Securities Australia Ltd Sell 1,471 (46) UBS Securities Australia Ltd Sell 1,4714 (1,637) UBS Securities Australia Ltd Sell 1,034 (1,637) UBS Securities Australia Ltd Sell 4,210 (1,637) UBS Securities Australia Ltd Sell 4,434 (1,637) UBS Securities Australia Ltd Sell 4,434 (1,637) UBS Securities Australia Ltd Sell 4,434 (4,63) UBS Securities Australia Ltd Sell 3,162 (4,63) UBS Securities Australia Ltd Sell 1,825 (4,63) UBS Securities Australia Ltd Sell 3,162 (4,63) UBS Securities Australia Ltd Sell 3,162 (4,63) UBS Securities Australia Ltd Sell 3,162 (4,23) UBS Securities Australia Ltd Sell 3,162 (4,23) UBS Securities Australia Ltd Sell 3,162 (4,23) UBS Securi	-Ju}-14	UBS Securities Australia Ltd	Sell	352	(36)	Ordinary
UBS Securities Australia Ltd Sell 2,545 (260) UBS Securities Australia Ltd Sell 14,714 (1,037) UBS Securities Australia Ltd Sell 1,0348 (1,057) UBS Securities Australia Ltd Sell 4,434 (1,037) UBS Securities Australia Ltd Sell 4,434 (459) UBS Securities Australia Ltd Sell 4,434 (459) UBS Securities Australia Ltd Sell 5,453 (459) UBS Securities Australia Ltd Sell 1,822 (185) UBS Securities Australia Ltd Sell 5,453 (5,57) UBS Securities Australia Ltd Sell 5,453 (5,60) UBS Securities Australia Ltd Sell 5,453 (5,60) UBS Securitie	-Jul-14	UBS Securities Australia Ltd	Self	1,291	(132)	Ordinary
UBS Securities Australia Ltd Sell 14,714 (1,503) UBS Securities Australia Ltd Sell 1,4714 (1,503) UBS Securities Australia Ltd Sell 1,003 (1,607) UBS Securities Australia Ltd Sell 4,210 (1,603) UBS Securities Australia Ltd Sell 4,41 (459) UBS Securities Australia Ltd Sell 4,44 (459) UBS Securities Australia Ltd Sell 4,41 (45) UBS Securities Australia Ltd Sell 3,162 (1,86) UBS Securities Australia Ltd Sell 3,162 (3,82) UBS Securities Australia Ltd Sell 3,162 (3,82) UBS Securities Australia Ltd Sell 5,129 (3,82) UBS Securities Australia Ltd Sell 3,162 (3,23) UBS Securities	-Jul-14	UBS Securities Australia Ltd	Sell	2.545	(260)	Ordinary
UBS Securities Australia Ltd Sell 14,714 (1,503) UBS Securities Australia Ltd Sell 4,214 (1,503) UBS Securities Australia Ltd Sell 4,494 (459) UBS Securities Australia Ltd Sell 245 (459) UBS Securities Australia Ltd Sell 245 (459) UBS Securities Australia Ltd Sell 241 (45) UBS Securities Australia Ltd Sell 3,162 (32) UBS Securities Australia Ltd Sell 3,162 (48) UBS Securities Australia Ltd Sell 3,162 (43) UBS Securities Australia Ltd Sell 5,453 (5,57) UBS Securities Australia Ltd Sell 5,453 (5,57) UBS Securities Australia Ltd Sell 5,453 (44) UBS Securities Australia Ltd Sell 5,453 (5,69) UBS Securities Australia Ltd Sell 5,453 (5,60) UBS Securities Australia Ltd Sell 5,91 (44) UBS Securities Australia Ltd <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>450</td> <td>(46)</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Sell	450	(46)	Ordinary
UBS Securities Australia Ltd Sell 10,348 (1,057) UBS Securities Australia Ltd Sell 4,210 (430) UBS Securities Australia Ltd Sell 1,093 (1033) UBS Securities Australia Ltd Sell 4,494 (459) UBS Securities Australia Ltd Sell 4,494 (459) UBS Securities Australia Ltd Sell 3,162 (185) UBS Securities Australia Ltd Sell 1,811 (185) UBS Securities Australia Ltd Sell 3,162 (357) UBS Securities Australia Ltd Sell 5,453 (5,857) UBS Securities Australia Ltd Sell 3,162 (444) UBS Securities Australia Ltd Sell 5,453 (5,677) UBS Securities Australia Ltd Sell 3,162 (424) UBS Securities Australia Ltd Sell 3,162 (424) UBS Securities Australia Ltd Sell 3,985 (407) UBS Securities Australia Ltd Sell 3,985 (407) UBS Securities Austr	-Jul-14	UBS Securities Australia Ltd	Sell	14 714	(1 503)	Ordinary
UBS Securities Australia Ltd Sell 4,710 (1/30) UBS Securities Australia Ltd Sell 1,009 (103) UBS Securities Australia Ltd Sell 245 (459) UBS Securities Australia Ltd Sell 245 (459) UBS Securities Australia Ltd Sell 3,162 (186) UBS Securities Australia Ltd Sell 1,822 (186) UBS Securities Australia Ltd Sell 245 (45) UBS Securities Australia Ltd Sell 3,162 (185) UBS Securities Australia Ltd Sell 3,162 (185) UBS Securities Australia Ltd Sell 3,162 (25) UBS Securities Australia Ltd Sell 3,162 (23) UBS Securities Australia Ltd Sell 3,162 (1,219) UBS Securities Australia Ltd Sell 3,965 (40) UBS Securities Australia Ltd Sell 3,965 (40) UBS Securities Australia Ltd Sell 3,965 (40) UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Itd	T-G-S	10 348	(1 057)	Ordinany
UBS Securities Australia Ltd Sell 1,009 (103) UBS Securities Australia Ltd Sell 4,494 (459) UBS Securities Australia Ltd Sell 245 (459) UBS Securities Australia Ltd Sell 4,41 (45) UBS Securities Australia Ltd Sell 1,822 (188) UBS Securities Australia Ltd Sell 3,162 (323) UBS Securities Australia Ltd Sell 2,453 (582) UBS Securities Australia Ltd Sell 5,453 (587) UBS Securities Australia Ltd Sell 5,453 (587) UBS Securities Australia Ltd Sell 5,453 (5,29) UBS Securities Australia Ltd Sell 3,162 (3,23) UBS Securities Australia Ltd Sell 5,29 (44) UBS Securities Australia Ltd Sell 3,162 (3,29) UBS Securities Australia Ltd Sell 3,162 (3,29) UBS Securities Australia Ltd Sell 3,985 (40) UBS Securities Australia Ltd <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>4 210</td> <td>(430)</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Sell	4 210	(430)	Ordinary
UBS Securities Australia Ltd Sell 4,950 (459) UBS Securities Australia Ltd Sell 245 (25) UBS Securities Australia Ltd Sell 3,162 (38) UBS Securities Australia Ltd Sell 3,162 (33) UBS Securities Australia Ltd Sell 3,162 (33) UBS Securities Australia Ltd Sell 3,162 (33) UBS Securities Australia Ltd Sell 5,729 (35) UBS Securities Australia Ltd Sell 5,729 (5,822) UBS Securities Australia Ltd Sell 5,437 (444) UBS Securities Australia Ltd Sell 5,437 (437) UBS Securities Australia Ltd Sell 5,437 (437) UBS Securities Australia Ltd Sell 3,162 (43) UBS Securities Australia Ltd Sell 3,985 (40) UBS Securities Australia Ltd Sell 3,985 (40) UBS Securities Australia Ltd Sell 3,985 (10) UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Ltd	<u></u>	000	(501)	Ordinary
UBS Securities Australia Ltd Sell 7.52 (75) UBS Securities Australia Ltd Sell 4.45 (48) UBS Securities Australia Ltd Sell 1,822 (186) UBS Securities Australia Ltd Sell 1,822 (185) UBS Securities Australia Ltd Sell 1,822 (185) UBS Securities Australia Ltd Sell 1,821 (185) UBS Securities Australia Ltd Sell 5,729 (5,822) UBS Securities Australia Ltd Sell 3,162 (2,57) UBS Securities Australia Ltd Sell 3,162 (2,23) UBS Securities Australia Ltd Sell 3,162 (3,23) UBS Securities Australia Ltd Sell 3,965 (4,07) UBS Securities Australia Ltd Sell 3,731 (5,225) UBS Securities Austra	-101-14	UBS Securities Australia I to	= 0	V V V	(450)	Ordinary Ordinary
UBS Securities Australia Ltd Sell 5.365 (5.48) UBS Securities Australia Ltd Sell 1,822 (4.85) UBS Securities Australia Ltd Sell 1,822 (4.85) UBS Securities Australia Ltd Sell 2,45 (5.87) UBS Securities Australia Ltd Sell 5,453 (5.27) UBS Securities Australia Ltd Sell 5,453 (5.27) UBS Securities Australia Ltd Sell 5,29 (5,44) UBS Securities Australia Ltd Sell 5,29 (44) UBS Securities Australia Ltd Sell 5,29 (40) UBS Securities Australia Ltd Sell 3,985 (40) UBS Securities Australia Ltd Sell 3,985 (40) UBS Securities Australia Ltd Sell 3,985 (40) UBS Securities Australia Lt	-1.11-14	I IBS Securities Australia (4)	100	tot't	(1004)	Couriery
UBS Securities Australia Ltd Sell 3,565 UBS Securities Australia Ltd Sell 441 (45) UBS Securities Australia Ltd Sell 1,62 (165) UBS Securities Australia Ltd Sell 1,61 (165) UBS Securities Australia Ltd Sell 3,729 (557) UBS Securities Australia Ltd Sell 5,453 (557) UBS Securities Australia Ltd Sell 3,162 (444) UBS Securities Australia Ltd Sell 3,162 (427) UBS Securities Australia Ltd Sell 5,291 (1,29) UBS Securities Australia Ltd Sell 3,162 (4,40) UBS Securities Australia Ltd Sell 3,162 (1,0) UBS Securities Australia Ltd Sell 3,162 (1,0) UBS Securities Australia Ltd Sell 3,162 (1,0) UBS Securities Australia Ltd Sell 3,985 (1,0) UBS Securities Australia Ltd Sell 3,985 (1,0) UBS Securities Australia Ltd Sell <td>11-11-17</td> <td>I IBC Countries Australia 14d</td> <td>1100</td> <td>C#2</td> <td>(67)</td> <td>Ordinary</td>	11-11-17	I IBC Countries Australia 14d	1100	C#2	(67)	Ordinary
UBS Securities Australia Ltd Sell 1,821 (165) UBS Securities Australia Ltd Sell 1,811 (185) UBS Securities Australia Ltd Sell 1,811 (185) UBS Securities Australia Ltd Sell 2,453 (5,822) UBS Securities Australia Ltd Sell 3,7591 (5,822) UBS Securities Australia Ltd Sell 4,347 (444) UBS Securities Australia Ltd Sell 3,162 (323) UBS Securities Australia Ltd Sell 3,162 (323) UBS Securities Australia Ltd Sell 3,965 (407) UBS Securities Australia Ltd Sell 8,956 (407) UBS Securities Australia Ltd Sell 8,956 (5,000) UBS Securities Australia Ltd Sell 8,743 (43) UBS Securities Austral	1111	LIDS Contribut Australia 14d	100	C05'C	(548)	Ordinary
UBS Securities Australia Ltd Sell 1.822 (186) UBS Securities Australia Ltd Sell 1.811 (185) UBS Securities Australia Ltd Sell 2.45 (25) UBS Securities Australia Ltd Sell 5,729 (587) UBS Securities Australia Ltd Sell 5,437 (444) UBS Securities Australia Ltd Sell 11,934 (1,219) UBS Securities Australia Ltd Sell 3,162 (523) UBS Securities Australia Ltd Sell 3,162 (44) UBS Securities Australia Ltd Sell 3,162 (407) UBS Securities Australia Ltd Sell 3,985 (10) UBS Securities Australia Ltd Sell 3,985 (10) UBS Securities Australia Ltd Sell 3,985 (10) UBS Securities Australia Ltd <td>-101-14</td> <td>UBS Securities Australia Ltd</td> <td>] Sell</td> <td>441</td> <td>(42)</td> <td>Ordinary</td>	-101-14	UBS Securities Australia Ltd] Sell	441	(42)	Ordinary
UBS Securities Australia Ltd Sell 3162 (323) UBS Securities Australia Ltd Sell 1,811 (185) UBS Securities Australia Ltd Sell 5,7291 (5,822) UBS Securities Australia Ltd Sell 5,453 (5,822) UBS Securities Australia Ltd Sell 3,162 (323) UBS Securities Australia Ltd Sell 2,20 (2,19) UBS Securities Australia Ltd Sell 3,962 (407) UBS Securities Australia Ltd Sell 3,1,541 (3,225) UBS Securities Australia Ltd Sell 3,1,541 (3,225) UBS Securities A	-Jul-14	UBS Securities Australia Ltd	Sell	1,822	(186)	Ordinary
UBS Securities Australia Ltd Sell 1811 (185) UBS Securities Australia Ltd Sell 5,453 (5,527) UBS Securities Australia Ltd Sell 5,453 (5527) UBS Securities Australia Ltd Sell 3,162 (223) UBS Securities Australia Ltd Sell 3,162 (223) UBS Securities Australia Ltd Sell 5,98 (4,4) UBS Securities Australia Ltd Sell 5,98 (4,0) UBS Securities Australia Ltd Sell 3,985 (4,0) UBS Securities Australia Ltd Sell 8,950 (5,000) UBS Securities Australia Ltd Sell 8,950 (5,000) UBS Securities Australia	-Jul-14	UBS Securities Australia Ltd	Sell	3,162	(323)	Ordinary
UBS Securities Australia Ltd Sell 54.5 (5.82) UBS Securities Australia Ltd Sell 5,7391 (5.822) UBS Securities Australia Ltd Sell 3,437 (444) UBS Securities Australia Ltd Sell 3,162 (323) UBS Securities Australia Ltd Sell 2,0 (2) UBS Securities Australia Ltd Sell 5,29 (2) UBS Securities Australia Ltd Sell 3,162 (2) UBS Securities Australia Ltd Sell 3,162 (2) UBS Securities Australia Ltd Sell 3,985 (407) UBS Securities Australia Ltd Sell 3,985 (407) UBS Securities Australia Ltd Sell 3,985 (40) UBS Securities Australia Ltd Sell 8,743 (5,000) UBS Securities Australia Ltd Sell 8,743 (5,000) UBS Securities Australia Ltd Sell 8,743 (6,325) UBS Securities Australia Ltd Sell 3,743 (6,325) UBS Securities Australia Ltd<	-Jul-14	UBS Securities Australia Ltd	Sell	1,811	(185)	Ordinary
UBS Securities Australia Ltd Sell 57,291 (5,652) UBS Securities Australia Ltd Sell 4,347 (444) UBS Securities Australia Ltd Sell 3,162 (323) UBS Securities Australia Ltd Sell 1,934 (1,219) UBS Securities Australia Ltd Sell 5,29 (44) UBS Securities Australia Ltd Sell 5,29 (407) UBS Securities Australia Ltd Sell 3,965 (407) UBS Securities Australia Ltd Sell 3,965 (407) UBS Securities Australia Ltd Sell 3,965 (407) UBS Securities Australia Ltd Sell 8,955 (5,000) UBS Securities Australia Ltd Sell 8,955 (5,000) UBS Securities Australia Ltd Sell 8,743 (5,84) UBS Securities Australia Ltd Sell 8,743 (5,84) UBS Securities Australia Ltd Sell 8,743 (5,84) UBS Securities Australia Ltd Sell 5,815 (5,84) UBS Securities Aus	-Jul-14	UBS Securities Australia Ltd	Sell	245	(22)	Ordinary
UBS Securities Australia Ltd Seli 5,453 (557) UBS Securities Australia Ltd Seli 4,347 (444) UBS Securities Australia Ltd Sell 3,162 (323) UBS Securities Australia Ltd Sell 5,29 (5,4) UBS Securities Australia Ltd Sell 5,29 (4,5) UBS Securities Australia Ltd Sell 3,985 (407) UBS Securities Australia Ltd Sell 3,985 (10) UBS Securities Australia Ltd Sell 3,985 (10) UBS Securities Australia Ltd Sell 3,985 (10) UBS Securities Australia Ltd Sell 3,154 (3,225) UBS Securities Australia Ltd Sell 3,154 (3,225) UBS Securities Australia Ltd Sell 3,154 (18) UBS Securities Australia Ltd </td <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>57.291</td> <td>(5,852)</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Sell	57.291	(5,852)	Ordinary
UBS Securities Australia Ltd Sell 4,347 (444) UBS Securities Australia Ltd Sell 1,62 (323) UBS Securities Australia Ltd Sell 2,0 (7,19) UBS Securities Australia Ltd Sell 3,762 (4,2) UBS Securities Australia Ltd Sell 3,962 (407) UBS Securities Australia Ltd Sell 48,950 (500) UBS Securities Australia Ltd Sell 31,541 (5,228) UBS Securities Australia Ltd Sell 31,541 (3,228) UBS Securities Australia Ltd Sell 3,743 (894) UBS Securities Australia Ltd Sell 3,743 (894) UBS Securities Australia Ltd Sell 3,743 (894) UBS Securities Australia Ltd Sell 5,743 (894) UBS Securities Australi	-Jul-14	UBS Securities Australia Ltd	la C	5 153	(55.7)	Ordinary
UBS Securities Australia Ltd Sell 3,145 (472) UBS Securities Australia Ltd Sell 11,934 (1,219) UBS Securities Australia Ltd Sell 520 (5) UBS Securities Australia Ltd Sell 5,965 (407) UBS Securities Australia Ltd Sell 3,965 (407) UBS Securities Australia Ltd Sell 6,393 (43) UBS Securities Australia Ltd Sell 6,393 (43) UBS Securities Australia Ltd Sell 48,950 (5,000) UBS Securities Australia Ltd Sell 8,43 (5,40) UBS Securities Australia Ltd Sell 31,541 (3,225) UBS Securities Australia Ltd Sell 31,541 (3,225) UBS Securities Australia Ltd Sell 5,815 (5,200) UBS Securities Australia Ltd Sell 5,634 (5,20) UBS Securities Australia Ltd Sell 5,634 (5,22) UBS Securities Australia Ltd Sell 5,634 (5,23) UBS Securities Au	-111-14	I IRS Securities Australia I to	100	2001	(100)	Cidifiary
UBS Securities Australia Ltd Sell 1,102.4 (12.9) UBS Securities Australia Ltd Sell 1,024 (1.2) UBS Securities Australia Ltd Sell 5.29 (5.4) UBS Securities Australia Ltd Sell 3,162 (4.3) UBS Securities Australia Ltd Sell 3,162 (4.3) UBS Securities Australia Ltd Sell 3,983 (4.0) UBS Securities Australia Ltd Sell 3,983 (4.0) UBS Securities Australia Ltd Sell 3,98 (1.0) UBS Securities Australia Ltd Sell 3,98 (1.0) UBS Securities Australia Ltd Sell 3,154 (3,225) UBS Securities Australia Ltd Sell 3,154 (3,225) UBS Securities Australia Ltd Sell 3,154 (3,225) UBS Securities Australia Ltd Sell 2,15 (2,225) UBS Securities Australia Ltd Sell 2,15 (2,225) UBS Securities Australia Ltd Sell 2,15 (3,225) UBS Securities Aust	10,17	TIPO Cognition Australia 14d	100	4,047	(444)	Ordinary
UBS Securities Australia Ltd Sell 1,544 (1,219) UBS Securities Australia Ltd Sell 5.29 (54) UBS Securities Australia Ltd Sell 5.29 (64) UBS Securities Australia Ltd Sell 3,162 (63) UBS Securities Australia Ltd Sell 39 (4) UBS Securities Australia Ltd Sell 89 (7) UBS Securities Australia Ltd Sell 8,89 (7) UBS Securities Australia Ltd Sell 8,815 (60) UBS Securities Australia Ltd Sell 8,743 (894) UBS Securities Australia Ltd Sell 5,743 (894) UBS Securities Australia Ltd Sell 5,743 (82) UBS Securities Australia Ltd	1	Upo Complete Australia Ltd	IIAC I	3,102	(323)	Ordinary
UBS Securities Australia Ltd Sell 520 (4) UBS Securities Australia Ltd Sell 529 (407) UBS Securities Australia Ltd Sell 3,965 (407) UBS Securities Australia Ltd Sell 6,393 (407) UBS Securities Australia Ltd Sell 6,393 (407) UBS Securities Australia Ltd Sell 6,393 (40) UBS Securities Australia Ltd Sell 48,950 (5,000) UBS Securities Australia Ltd Sell 3,143 (894) UBS Securities Australia Ltd Sell 3,143 (894) UBS Securities Australia Ltd Sell 3,143 (894) UBS Securities Australia Ltd Sell 5,155 (5,20) UBS Securities Australia Ltd Sell 5,09 (5,22) UBS Securities Australia Ltd Sell 5,09 (5,2) UBS Securities Australia Ltd Sell 5,09 (5,2) UBS Securities Australia Ltd Sell 5,09 (5,2) UBS Securities Australia Ltd	41-15-14-14-14-14-14-14-14-14-14-14-14-14-14-	Opposedining Australia Ltd	Sell	11,934	(1,219)	Ordinary
UBS Securities Australia Ltd Sell 529 (54) UBS Securities Australia Ltd Sell 3,62 (32) UBS Securities Australia Ltd Sell 3,162 (32) UBS Securities Australia Ltd Sell 3,98 (40) UBS Securities Australia Ltd Sell 3,87 (4) UBS Securities Australia Ltd Sell 3,58 (10) UBS Securities Australia Ltd Sell 3,541 (5,000) UBS Securities Australia Ltd Sell 3,73 (894) UBS Securities Australia Ltd Sell 3,73 (894) UBS Securities Australia Ltd Sell 3,73 (894) UBS Securities Australia Ltd Sell 3,73 (18) UBS Securities Australia Ltd Sell 3,93 (18) UBS Securities Australia Ltd Sell 3,93 (31)	1-10-	UBS Securities Australia Ltd	Seil	20	(2)	Ordinary
UBS Securities Australia Ltd Sell 3,985 (407) UBS Securities Australia Ltd Sell 3,162 (333) UBS Securities Australia Ltd Sell 6,393 (43) UBS Securities Australia Ltd Sell 98 (4) UBS Securities Australia Ltd Sell 8,250 (5000) UBS Securities Australia Ltd Sell 8,743 (634) UBS Securities Australia Ltd Sell 8,743 (634) UBS Securities Australia Ltd Sell 8,743 (834) UBS Securities Australia Ltd Sell 8,743 (834) UBS Securities Australia Ltd Sell 215 (22) UBS Securities Australia Ltd Sell 5,93 (52) UBS Securities Australia Ltd Sell 5,93 (52) UBS Securities Australia Ltd Sell 5,93 (52) UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	-Jul-14	UBS Securities Australia Ltd	Sell	529	(54)	Ordinary
UBS Securities Australia Ltd Sell 3.1 G2 (333) UBS Securities Australia Ltd Sell 6,593 (45) UBS Securities Australia Ltd Sell 39 (10) UBS Securities Australia Ltd Sell 49,550 (5,000) UBS Securities Australia Ltd Sell 31,541 (3,225) UBS Securities Australia Ltd Sell 8,43 (894) UBS Securities Australia Ltd Sell 8,13 (824) UBS Securities Australia Ltd Sell 8,13 (824) UBS Securities Australia Ltd Sell 215 (222) UBS Securities Australia Ltd Sell 5,13 (82) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 303 (31)	-Jul-14	UBS Securities Australia Ltd	Sell	3,985	(407)	Ordinary
UBS Securities Australia Ltd Sell 6,393 (653) UBS Securities Australia Ltd Sell 39 (4) UBS Securities Australia Ltd Sell 98 (10) UBS Securities Australia Ltd Sell 48,950 (5,000) UBS Securities Australia Ltd Sell (5,000) UBS Securities Australia Ltd Sell (3,255) UBS Securities Australia Ltd Sell (2,225) UBS Securities Australia Ltd Sell (2,22) UBS Securities Australia Ltd Sell (5,22) UBS Securities Australia Ltd Sell (5,22) UBS Securities Australia Ltd Sell (5,22) UBS Securities Australia Ltd Sell (5,23) UBS Securities Australia Ltd Sell (5,23) UBS Securities Australia Ltd Sell (5,23) UBS Securities Australia Ltd Sell (3,3) UBS Securities Australia Ltd Sell (3,1)	-Jul-14	UBS Securities Australia Ltd	Sell	3,162	(323)	Ordinary
UBS Securities Australia Ltd Sell 39 (4) UBS Securities Australia Ltd Sell 48,950 (5,000) UBS Securities Australia Ltd Sell 48,950 (5,000) UBS Securities Australia Ltd Sell 31,541 (3,225) UBS Securities Australia Ltd Sell 8,743 (894) UBS Securities Australia Ltd Sell 215 (22) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	-Jul-14	UBS Securities Australia Ltd	Sell	868.9	(653)	Ordinary
UBS Securities Australia Ltd Sell 48.950 (10) UBS Securities Australia Ltd Sell 48.950 (5,000) UBS Securities Australia Ltd Sell 5.815 (5,000) UBS Securities Australia Ltd Sell 8.743 (894) UBS Securities Australia Ltd Sell 215 (22) UBS Securities Australia Ltd Sell (52) (52) UBS Securities Australia Ltd Sell (52) (62) UBS Securities Australia Ltd Sell (62) (18) UBS Securities Australia Ltd Sell (76) (18) UBS Securities Australia Ltd Sell 303 (31)	-Jul-14	UBS Securities Australia Ltd	las	95	(4)	Ordina
UBS Securities Australia Ltd Sell 48,950 (5,000) UBS Securities Australia Ltd Sell 5,815 (5,000) UBS Securities Australia Ltd Sell 31,541 (3,255) UBS Securities Australia Ltd Sell 8,743 (894) UBS Securities Australia Ltd Sell 215 (22) UBS Securities Australia Ltd Sell 579 (52) UBS Securities Australia Ltd Sell 579 (52) UBS Securities Australia Ltd Sell 579 (18) UBS Securities Australia Ltd Sell 303 (13) UBS Securities Australia Ltd Sell 303 (31)	-1ul-14	UBS Securities Australia I to	Sol	80	(1)	Ordinary
UBS Securities Australia Ltd Sell 5,100 UBS Securities Australia Ltd Sell 31,541 (3,125) UBS Securities Australia Ltd Sell 8,143 (894) UBS Securities Australia Ltd Sell 8,143 (894) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 5,09 (18) UBS Securities Australia Ltd Sell 303 (31)	11.14	A DC Conwision A section 14	=======================================	40.070	(01)	Ç Gillary
UBS Securities Australia Ltd Sell 3,515 (594) UBS Securities Australia Ltd Sell 31,541 (3,255) UBS Securities Australia Ltd Sell 2,15 (22) UBS Securities Australia Ltd Sell 5,09 (52) UBS Securities Australia Ltd Sell 176 (18) UBS Securities Australia Ltd Sell 176 (18) UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	1	Libe Committee Australia Lice	naci	46,930	(Onn's)	Ordinary
UBS Securities Australia Ltd Sell 8,1541 (3,225) UBS Securities Australia Ltd Sell 8,743 (894) UBS Securities Australia Ltd Sell 215 (22) UBS Securities Australia Ltd Sell 579 (52) UBS Securities Australia Ltd Sell 176 (18) UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	+	Obsistenties Australia Ltd	Sell	518,5	(594)	Ordinary
UBS Securities Australia Ltd Seil 8,743 (894) UBS Securities Australia Ltd Sell 215 (22) UBS Securities Australia Ltd Sell 509 (52) UBS Securities Australia Ltd Sell 76 (18) UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	-Jul-14	UBS Securities Australia Ltd	Sell	31,541	(3,225)	Ordinary
UBS Securities Australia Ltd Sell 215 (22) UBS Securities Australia Ltd Sell 509 (52) UBS Securities Australia Ltd Sell 176 (18) UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	-Jul-14	UBS Securities Australia Ltd	Sell	8,743	(834)	Ordinary
UBS Securities Australia Ltd Sell 509 (52) UBS Securities Australia Ltd Sell 176 (18) UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	-Jul-14	UBS Securities Australia Ltd	Seil	215	(22)	Ordinary
UBS Securities Australia Ltd Sell 176 (18) UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	-Jul-14	UBS Securities Australia Ltd	Sell	605	(52)	Ordinary
UBS Securities Australia Ltd Sell 303 (31) UBS Securities Australia Ltd Sell 303 (31)	-Jui-14	UBS Securities Australia Ltd	IleS	176	(18)	Ordinary
UBS Securities Australia Ltd Sell 303 (31)	-lul-14	LBS Securities Australia Ltd	= 0	303	(24)	
UDS Securities Australia Ltd (31)	6.4.3.4	DO DUDICON LOCAL MARIA				100000
	- 111-14	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				Ordinary

UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Sell	ctc	(32)	
UBS Securities Australia Ltd UBS Securities Australia Ltd		CIC		Ordinary
rrities Australia Ltd	Sell	8/6	(100)	
	Sell	5,956	(609)	
urities Australia Ltd	Sell	929	(64)	Ordinary
UBS Securities Australia Ltd	Seil	352	(36)	
UBS Securities Australia Ltd	Seli	3,482	(356)	
UBS Securities Australia Ltd	Seli	5,633	(576)	Ordinary
UBS Securities Australia Ltd	Sell	5,985	(612)	
urities Australia Ltd	Sell	9.428	(964)	
JRS Securities Australia I to	Cell	120	(30)	l
unities Australia I to	3	CZ 2	(220)	l
DC Contribut Australia 14d	ll-S	3110	(100)	
Company of the compan	===	0//7	(204)	Ordinary
Securities Australia Ltd	Sell	6,132	(627)	
curties Australia Ltd	Sett	910	(93)	Ordinary
UBS Securities Australia Ltd	Self	841	(98)	Ordinary
Securities Australia Ltd	Sell	645	(99)	Ordinary
UBS Securities Australia Ltd	38	2,582	(264)	
ourities Australia I Id	195	2 504	(256)	Judgipao
1 IDC Co. without A contraction 1 and	1143	100/2	(202)	
COLUMN MOST AND THE	nac .	5,22,5	(354)	
curities Australia Ltd	Sei	9,438	(965)	
UBS Securities Australia Ltd	Sell	6,885	(104)	Ordinary
JBS Securities Australia Ltd	Sell	9.135	(934)	
JBS Securities Australia Ltd	3	787 0	(285)	
PS Securities Australia 140	= 0	22,720	(205.0)	
Carrier Australia 144		0000	(2000)	
Securities Australia Ltu	130	8,0/8	(979)	1
UBS Securities Australia Ltd	Sett	3,932	(402)	
UBS Securities Australia Ltd	Sell	4,919	(203)	Ordinary
UBS Securities Australia Ltd	Sell	1,751	(179)	
UBS Securities Australia Ltd	les	1828	(33)	L
IRS Securities Australia I to	Sall	212	(00)	Crains
Author Australia 14d	30	000	(20)	Ordinal y
DO Consisting Australia Land	150	-60	(/0)	Clumary
Culties Australia Ltd	120	4,040	(co+)	Ordinary
JBS Securities Australia Ltd	Sell	3,159	(323)	Ordinary
JBS Securities Australia Ltd) set	5,45/	(558)	Ordinary
JBS securities Australia Ltd	Sell	2,631	(569)	Ordinary
JBS Securities Australia Ltd	Sell	3,922	(401)	Ordinary
UBS Securities Australia Ltd	Sell	26,426	(2,702)	Ordinary
ecurities Australia Ltd	Sell	26,064	(3,665)	Ordinary
UBS Securities Australia Ltd	llay	7 599	(777)	Ordinary
185 Sequeties Australia 14d	[e]	50031	(5,112)	Ordinary
Principal Commission C		120,02	(2) (4)	Cicinaly
curries Australia Ltd	Sell	1,144	(31%)	Ordinary
Securities Australia Ltd	Seli	3,159	(323)	Ordinary
UBS Securities Australia Ltd	Sell	3,159	(323)	Ordinary
UBS Securities Australia Ltd	Sell	3 159	(505)	Ordinary
A retroit Australia 144	Coll	35.15	(070)	Ordinary
Obstacting Australia Liu	100	6570	(040)	Ordinary
curities Australia Ltd	Sell	1,848	(189)	Ordinary
UBS Securities Australia Ltd	Seil	12,743	(1,303)	Ordinary
UBS Securities Australia Ltd	Sell	4.548	(465)	Ordinary
ourities Australia I to	llay.	3 345	(CVE)	Ordinativ
Origina Australia I tol	Coll	7000	(27.5)	Caringo
Collines Australiana Lica	270	#66°°	(347)	Oldinary
Obsider Australia Lid	Sell	4,039	(413)	Ordinary
curities Australia Ltd	Sell	12,646	(1,293)	Ordinary
JBS Securities Australia Ltd	Sell	16,205	(1,657)	Ordinary
IBS Securities Australia I td	Sell	21 672	(2.216)	Ordinary
IBS Securities Australia Ltd	lo S	9090	(000)	Contract of
The second secon	= 3	000'0	(000)	o o
ope securities Australia Ltd	ner o	000,1	(061)	Ordinary
curities Australia Ltd	Sell	ξρο'/	(8//)	Ordinary
UBS Securities Australia Ltd	Sell	225	(23)	Ordinary
UBS Securities Australia Ltd	Sell	587	(09)	Ordinary
curities Australia Ltd	Sell	23,335	(2,386)	Ordinary
JBS Securities Australia Ltd	Sell	1.467	(150)	Ordinary
180 Counties Australia 14d	3	2012	12.50	Ordings
Cultures Australia Ciu	361	710'/	(/ / /)	Ordinary
UBS Securities Australia Ltd	Sell	13,047	(1,334)	Ordinary
ourities Australia Ltd	lleS	1.105	(113)	Ordinary
ODS Securities Australia Etc		501,1	(01.)	Oldinary

Constructs Assistate Ltd SM 2,222 (2,23) 1065 Scourties Assistate Ltd SMI 2,222 (2,28) 1065 Scourties Assistate Ltd SMI 2,222 (2,28) 1065 Scourties Assistate Ltd SMI 10,794 (1,100) 1065 Scourties Assistate Ltd SMI 10,704 (2,28) 1065 Scourties Assistate Ltd SMI 10,704 (1,00) 1065 Scourties Assistate Ltd SMI 10,704 (2,28) 1065 Scourties Assistate Ltd SMI 10,704 (1,00) 1065 Scourties Assis		eel eel eel sock returned Stock returned	2,328 2,328 19,787 10,787 10,734 13,960 10,561 11,786 13,786 13,786 13,786 13,786 13,786 13,786 13,786 14,001 17,885 18,178 18,1	(5)	Ordinary Ordinary
185 Scorrifes Australia Ltd Seri 1,2,28 (1,29) 185 Scorrifes Australia Ltd Seri 1,2,28 (1,29) 185 Scorrifes Australia Ltd Sinch returned NA (3,000,000) 185 Scorrifes Australia Ltd Shock returned NA (3,000,000) 185 Scorrifes Australia Ltd Buy 2,3,28 (1,100) 185 Scorrifes Australia Ltd Buy 2,3,000 (4,101) 185 Scorrifes Australia Ltd Buy 2,3,000 (4,102) 185 Scorrifes Australia Ltd Buy 2,3,000 (4,101) 185 Scorrifes Australia Ltd Buy 2,3,000 (4,102) 185 Scorrifes Australia Ltd Buy 2,3,000 (4,200) 185 Scorrifes Australia Ltd Buy 2,3,000 (4,200) 185 Scorrifes Australia Ltd Buy 2,3,000 (4,200)		Sock returned Stock r	2,328 3,199 NAN NAN NAN NAN NAN 10,784 10,561	(3)	Ordinary Ordinary
USS Securities Australia Ltd State		See I Stock returned	10,786 10,786 10,786 10,786 10,789 10,7,989 10,7,989 10,7,989 10,7,989 10,7,989 10,7,989 10,010 11,8,017 112,904 112,904 112,904	(3)	Ordinary Ordinary
(1) Committee Australia Lud Serial (1) 758 (1) 100 (105 Scounties Australia Lud Social remained 10,778 (1,100) (105 Scounties Australia Lud Stock returned NAM (3,000,000) (105 Scounties Australia Lud Boy (1,000) (1,000) (105 Scounties Australia Lud Boy (1,000) (1,		ell in the control of	10,758 NA 13960 10,758 13,734 13,561 10,561 13,789 13,789 13,789 14,001 25,74 25,778 25,778 25,778 25,778 26,778 27,786 28,274 285,778 28	(5)	Ordinary Ordinary
UBS Securities Australia Ltd Securities Australia Ltd Sinck returned IV/D NA (1,10) UBS Securities Australia Ltd Shock returned NAM (5,000,000) UBS Securities Australia Ltd Shock returned NAM (5,000,000) UBS Securities Australia Ltd Buy 7,778 4,778 UBS Securities Australia Ltd Buy 7,734 4,101 UBS Securities Australia Ltd Buy 7,574 4,101 UBS Securities Australia Ltd Buy 7,574 4,101 UBS Securities Australia Ltd Buy 7,573 4,503 UBS Securities Australia Ltd Buy 8,204 1,101 UBS Securities Australia Ltd Buy 2,527 2,597 UBS Securities Australia Ltd Buy 2,597 4,503 UBS Securities Australia Ltd Buy 2,597 2,597 UBS Securities Australia Ltd Buy 2,597 2,597 UBS Securities Australia Ltd Buy 2,597 2,597 UBS Securities Australia Ltd Buy 2,597 <td< td=""><td></td><td>Stock returned Stock returned Stock returned Noy Noy Noy Noy Noy Noy Noy Noy Noy Noy</td><td>10,788 10,5,734 105,611 105,611 105,611 137,887 105,611 18,007 29,2178 18,017 18,017 18,017 18,017 18,017 18,017 18,017 18,017 18,017 29 29 20 20 20 20 20 21,395 21</td><td>(3)</td><td>Ordinary Ordinary /td></td<>		Stock returned Stock returned Stock returned Noy Noy Noy Noy Noy Noy Noy Noy Noy Noy	10,788 10,5,734 105,611 105,611 105,611 137,887 105,611 18,007 29,2178 18,017 18,017 18,017 18,017 18,017 18,017 18,017 18,017 18,017 29 29 20 20 20 20 20 21,395 21	(3)	Ordinary Ordinary
USS Securities Australia Ltd Soci returned NA (3,000,000)		Stock returned Stock	NA NA NA NA NA NA NA NA NA NA NA NA NA N	(5)	Ordinary Ordinary
UBS Securities Australia Ltd Stock returned NAM (3,000,000) UBS Securities Australia Ltd Stock returned NAM (5,000,000) UBS Securities Australia Ltd Buy 7,778 4,000 UBS Securities Australia Ltd Buy 7,778 4,000 UBS Securities Australia Ltd Buy 7,778 4,101 UBS Securities Australia Ltd Buy 7,778 4,101 UBS Securities Australia Ltd Buy 7,778 4,101 UBS Securities Australia Ltd Buy 1,102 1,103 UBS Securities Australia Ltd Buy 2,567 1,699 UBS Securities Australia Ltd Buy 2,577 1,699 UBS Securities Australia Ltd Buy 2,577 2,597 UBS Securities Australia Ltd Buy 2,597 2,597 UBS Securitie		Stock returned Stock	NA NA NA NA NA NA NA NA NA NA NA NA NA N		Ordinary Ordinary
UBS Securities Australia Ltd Stock returned NAA (7,294) UBS Securities Australia Ltd Buy 73,734 7,778 UBS Securities Australia Ltd Buy 7,934 7,778 UBS Securities Australia Ltd Buy 7,934 7,778 UBS Securities Australia Ltd Buy 7,934 7,778 UBS Securities Australia Ltd Buy 1,172 14,132 UBS Securities Australia Ltd Buy 1,132 1,137 UBS Securities Australia Ltd Buy 1,137 1,137 UBS Securities Australia Ltd Buy 1,138 1,136 UBS Securities Australia Ltd Buy 1,138 1,139 UBS Securities Australia Ltd Buy 1,139 1,136 UBS Securities Australia Ltd Buy 1,139 1,136 UBS Securities Australia Ltd Buy 1,139 1,136 UBS Securities Australia Ltd Buy 1,235 2,397 UBS Securities Australia Ltd Buy 1,235 2,397 UBS Securities Australia Ltd </td <td></td> <td>sick returned Stock r</td> <td>NA 75,734 43,960 7,78,736 105,611 137,887 105,611 7,816 6,826 885,274 285,7178 NA 7,853 12,395 23,395 23,395 23,395 23,395 23,395 23,395 23,395 23,395 20,017 18,017 20 21,292 23,395 24,092 25,001 26,001 27,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001</td> <td>(5,1</td> <td>Ordinary Ordinary /td>		sick returned Stock r	NA 75,734 43,960 7,78,736 105,611 137,887 105,611 7,816 6,826 885,274 285,7178 NA 7,853 12,395 23,395 23,395 23,395 23,395 23,395 23,395 23,395 23,395 20,017 18,017 20 21,292 23,395 24,092 25,001 26,001 27,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001 28,001	(5,1	Ordinary Ordinary
UBS Securities Australia Ltd Shock returned NA (5,000,000) UBS Securities Australia Ltd Buy 7,778 4,000		Stock returned May Ma	75,734 43,960 105,611 117,887 40,001 7,816 6,826 885,274 7,816 18,178 18	(5,0	Ordinary Ordinary
UBS Securities Australia Ltd Buy 75/734 77/8 UBS Securities Australia Ltd Sell 15/8 4/506 UBS Securities Australia Ltd Sell 1/500 4/506 UBS Securities Australia Ltd Sell 1/500 1/500 UBS Securities Australia Ltd Sell 7.8616 (820) UBS Securities Australia Ltd Sell 7.8616 (820) UBS Securities Australia Ltd Sell 7.8616 (820) UBS Securities Australia Ltd Buy 2.85178 2.846 UBS Securities Australia Ltd Buy 2.85178 2.894 UBS Securities Australia Ltd Buy 2.895 2.897 UBS Securities Australia Ltd Buy 2.29 2.397 UBS Securities Australia Ltd Buy 2.29 2.397 UBS Securities Australia Ltd Buy 2.25 2.897 UBS Securities Australia Ltd Buy 2.25 2.897 UBS Securities Australia Ltd Buy 2.25 2.25 UBS Securities Australia Ltd		Nay Nay Nay Nay Nay Nay Nay Nay	75,74 43,960 43,960 105,611 107,887 40,001 7,816 6,826 6,826 6,826 1,827 1810 7,816 7,816 7,816 7,816 7,816 7,816 7,816 7,816 7,816 7,817 8,23 7,701 7,816 7,817 8,23 7,701 7,816 7,817 8,23 7,817 8,23 7,701 7,817 8,23 7,701 7,817 8,23 7,701 7,817 8,23 7,23 7,23 7,23 7,23 7,23 7,23 7,23 7		Ordinary Ordinary
UBS Securities Australia Ltd Buy 4.95 960 4.50 66 UBS Securities Australia Ltd Sell 7,816 (10,22) UBS Securities Australia Ltd Sell 1,020 (10,132) UBS Securities Australia Ltd Sell 7,816 (10,032) UBS Securities Australia Ltd Buy 2,527 1,410 UBS Securities Australia Ltd Buy 2,537 1,616 UBS Securities Australia Ltd Buy 2,537 1,626 UBS Securities Australia Ltd Buy 2,537 1,626 UBS Securities Australia Ltd Buy 2,537 2,597 UBS Securities Australia Ltd Buy 2,597 2,597 UBS Securities Australia Ltd Buy 2,598 2,597 UBS Securities Australia Lt		bay bay bay bay lefel lefel lefel luy bay bay bay bay bay bay bay bay bay ba	43,960 105,611 137,887 40,001 7,816 85,274 285,274 285,274 17,853 17,010 18,017 29 29 29 20 20 20 20 20 20 20 20 20 20 20 20 20		Ordinary Ordinary
UBS Securities Australia Ltd Sell 7,989 (81) UBS Securities Australia Ltd Boy 10,789 (82) UBS Securities Australia Ltd Sell 10,789 (14,132) UBS Securities Australia Ltd Sell 7,816 (820) UBS Securities Australia Ltd Sell 7,816 (820) UBS Securities Australia Ltd Boy 10,827 1,700 UBS Securities Australia Ltd Boy 28,718 1,500 UBS Securities Australia Ltd Boy 28,537 3,949 UBS Securities Australia Ltd Boy 2,937 3,949 UBS Securities Australia Ltd Boy 2,937 3,949 UBS Securities Australia Ltd Boy 2,336 2,597 UBS Securities Australia Ltd		eell leel leel leel leel loub luby luby luby luby luby luby luby lu	7,389 137,887 137,887 7,816 6,826 6,826 6,827 255,78 70,70 17,853 18,07 18,07 10,09 29 29 20 20 20 20 20 20 20 20 20 20 20 20 20		Ordinary Ordinary
UBS Securities Australia Ltd Sell 105,551 (10,520) UBS Securities Australia Ltd Sell 40,501 (4,101) UBS Securities Australia Ltd Sell 40,001 (4,101) UBS Securities Australia Ltd Buy 50,001 (4,101) UBS Securities Australia Ltd Buy 20,001 2,200 UBS Securities Australia Ltd Buy 2,395 2,397 UBS Securities Australia Ltd Buy 2,204 98,20 UBS Securities A		boby boby boby boby boby boby boby boby	105,611 137,887 1807 6,826 385,274 285,178 14,085 14,085 29 29 23,395 33,992 20 20 21,395 20 20 21,395 20 20 21,395 20 20 20 20 20 20 20 20 20 20 20 20 20		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Bay 17,329 1,4320 UBS Securities Australia Ltd Self 7,001 (4,101) UBS Securities Australia Ltd Self 7,001 (4,101) UBS Securities Australia Ltd Buy 386,2778 (700) UBS Securities Australia Ltd Buy 25,5178 (26,149) UBS Securities Australia Ltd Buy 23,592 2,049 UBS Securities Australia Ltd Buy 23,392 2,049 UBS Securities Australia Ltd Buy 23,392 2,397 UBS Securities Australia Ltd Buy 2,394 3,792 UBS Securities Austra		buy leal leal leal leal leal leal leal lea	137,887 40,001 7,816 6,826 88,274 255,178 18,017 18,017 29 29 29 37,010		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Self 40,001 (4,101) UBS Securities Australia Ltd Buly 58,274 30,490 UBS Securities Australia Ltd Buly 58,274 30,490 UBS Securities Australia Ltd Buly 25,049 1,040 UBS Securities Australia Ltd Buly 2,049 2,049 UBS Securities Australia Ltd Buly 2,957 4,503 UBS Securities Australia Ltd Buly 2,957 4,200 UBS Securities Australia Ltd Buly 2,959 2,957 UBS Securities Australia Ltd Buly 2,950 2,957 UBS Securities Australia Ltd Buly 2,950 2,957 UBS Securities Australia Ltd Buly 2,049 1,1568 UBS Securities Australia Ltd Buly 2,049 1,1568 UBS Securities Australia Ltd Buly 2,050 2,297 UBS Securities Australia Ltd Buly 2,049 1,1568 UBS Securities Australia Ltd Buly 2,050 2,297 UBS Securities Au		eell hande h	17,000 40,001 1,000 1,000 1,000 111,004 1,000		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
(BS Securities Australia Ltd Sell 7,3161 (800) (BS Securities Australia Ltd Sell 7,3161 (800) (BS Securities Australia Ltd Buy 385,274 39,493 (BS Securities Australia Ltd Buy 385,277 30,03 (BS Securities Australia Ltd Buy 25,577 3,03 (BS Securities Australia Ltd Buy 23,597 3,03 (BS Securities Australia Ltd Buy 23,597 3,03 (BS Securities Australia Ltd Buy 23,597 3,792 (BS Securities Australia Ltd Buy 23,597 3,597 (BS Securities Australia Ltd Buy 23,597 3,597 (BS Securities Australia Ltd Buy 23,597 3,597 (BS Securities Australia Ltd Buy 23,598 2,597 (BS Securities Australia L		eel leel leel leel leel leel leel leel	40,000 6,826 385,274 285,178 NA 47,853 1017 18,007 29 23,395 37,010 40,992 20 20 21,395 33,992 20 20 21,395 33,992 20 20 21,395 33,992 20 20 20 20 20 20 20 20 20 20 20 20 20		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd SH (AB 10) (AB 10) UBS Securities Australia Ltd Buy 385,274 39,493 UBS Securities Australia Ltd Buy 25,499 36,493 UBS Securities Australia Ltd Buy 25,049 36,493 UBS Securities Australia Ltd Buy 23,395 2,993 UBS Securities Australia Ltd Buy 23,395 2,997 UBS Securities Australia Ltd Buy 2,995 2,997 UBS Securities Australia Ltd Buy 2,096 1,184 UBS Securities Australia Ltd Buy 2,096 1,184 UBS Securities Australia Ltd Buy 2,096 2,297 UBS Securities Australia Ltd Buy 2,096 1,275 UBS Securities Australia Ltd Buy 2,096 1,276 UBS Securities Australia Ltd		e el l'action de l	6.826 885,274 255,178 8017 8017 18,017 29 29 20 37,010 40,992 23,395 23,395 24,992 23,395 24,992 27,010 27,01		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Sell 6,825 (700) UBS Securities Australia Ltd Sell 386,277 34,939 UBS Securities Australia Ltd Buy 251,78 (26,149) UBS Securities Australia Ltd Buy 20,70 3 UBS Securities Australia Ltd Buy 23,397 4,603 UBS Securities Australia Ltd Buy 23,397 4,603 UBS Securities Australia Ltd Buy 23,397 4,603 UBS Securities Australia Ltd Buy 23,397 4,500 UBS Securities Australia Ltd Buy 23,397 4,200 UBS Securities Australia Ltd Buy 23,397 4,200 UBS Securities Australia Ltd Buy 23,397 4,200 UBS Securities Australia Ltd Buy 2,51 5,397 UBS Securities Australia Ltd Buy 2,521 3,57 UBS Securities Australia Ltd Buy 2,73 4,260 UBS Securities Australia Ltd Buy 2,73 4,260 UBS Securities Australia Ltd		augh eel mine bringine C eel mine bringine C to by to	385,274 255,178 255,178 100 18,017 29 23,395		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Bluy 35,74 39,439 UBS Securities Australia Ltd Bluy 26,491 26,493 UBS Securities Australia Ltd Buy 2,049 2,049 UBS Securities Australia Ltd Buy 2,23 2,337 UBS Securities Australia Ltd Buy 2,03 2,03 UBS Securities Australia Ltd Buy		buy buy triner into Derivative C buy buy buy buy buy buy buy buy buy buy	385,274 25,178 7,185 7,185 18,017 29 29 29 29 23,395 37,010 40,992 29 3,395 29 3,395 29 3,395 29 3,395 20 112,904		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Self CSS 178 CSS 178 CSS 149 UBS Securities Australia Ltd Buy 2043 4,033 4,043 UBS Securities Australia Ltd Buy 2,29 3 UBS Securities Australia Ltd Buy 23 4,200 UBS Securities Australia Ltd Buy 23 2,30 UBS Securities Australia Ltd Buy 2,30 2,30 UBS Securities Australia Ltd Buy 2,23 2,30 UBS Securities Australia Ltd Buy 2,23 2,30 UBS Securities Australia Ltd Buy 1,1,2,50 1,1,2,63 UBS Securities Australia Ltd Buy 2,2,1 2,2 UBS Securities Australia Ltd Buy 1,1,3,50 1,1,2,6 UBS Securities Australia Ltd		ell mare into Derivative C int	255,178 NAA 47,853 18,017 12,92 23,395 37,010 40,992 23,395 37,010 112,904		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Either into Derivative AVA 2,049 UBS Securities Australia Ltd Buy 1,057 1,1246 UBS Securities Australia Ltd Buy 23 1,397 UBS Securities Australia Ltd Buy 23 2,397 UBS Securities Australia Ltd Buy 2,395 2,397 UBS Securities Australia Ltd Buy 2,20 2,397 UBS Securities Australia Ltd Buy 2,20 2,20 UBS Securities Australia Ltd Buy 2,20 2,20 UBS Securities Australia Ltd Buy 2,20 2,20 UBS Securities Australia Ltd Buy 5,7,935 5,936 UBS Securities Australia Ltd Buy 1,1,563 1,1,663 UBS Securities Australia Ltd Buy 1,5,937 1,1,663 UBS Securities Australia Ltd		riter Into Derivative C buy buy buy buy buy buy buy buy buy buy	47,853 47,853 18,017 23,325 23,325 23,395 23,395 23,395 23,395 23,395 20 20 20 20 20 20 20 20 20 20 20 20 20		Options Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 4,933 4,933 UBS Securities Australia Ltd Buy 22 3 UBS Securities Australia Ltd Buy 22 3 UBS Securities Australia Ltd Buy 23 3 UBS Securities Australia Ltd Buy 23 39 UBS Securities Australia Ltd Buy 23 23 UBS Securities Australia Ltd Buy 20 22 UBS Securities Australia Ltd Buy 57,932 2,936 UBS Securities Australia Ltd Buy 5,584 982 UBS Securities Australia Ltd Buy 5,584 982 UBS Securities Australia Ltd Buy 5,584 982 UBS Securities Australia Ltd Buy 5,584 9,584 UBS Securities Australia Ltd Buy 5,510 1,730		buy buy buy buy buy buy buy buy buy buy	47,853 18,017 29 29 37,010 40,992 23,395 28,395 29,395 29,395 29,395 20,		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Bby 18,017 1,846 UBS Securities Australia Ltd Buy 29 3 UBS Securities Australia Ltd Buy 2395 2,397 UBS Securities Australia Ltd Buy 23,395 2,397 UBS Securities Australia Ltd Buy 2,204 1,188 UBS Securities Australia Ltd Buy 2,204 1,188 UBS Securities Australia Ltd Buy 2,204 1,188 UBS Securities Australia Ltd Buy 2,231 2,397 UBS Securities Australia Ltd Buy 1,184 4,094 UBS Securities Australia Ltd Buy 1,184 4,094 UBS Securities Australia Ltd Buy 1,1256 1,184 UBS Securities Australia Ltd Buy 1,1256 1,184 UBS Securities Australia Ltd <t< td=""><td>1 </td><td>And And And And And And And And And And</td><td>18,017 29 23,395 37,010 40,992 23,395 23,395 39 112,904 112,904</td><td></td><td>Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary</td></t<>	1	And And And And And And And And And And	18,017 29 23,395 37,010 40,992 23,395 23,395 39 112,904 112,904		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 2.9 3 UBS Securities Australia Ltd Buy 2.3950 2.397 UBS Securities Australia Ltd Buy 2.7335 2.395 UBS Securities Australia Ltd Buy 1.15.56 1.184 UBS Securities Australia Ltd Buy 5.739 2.29 UBS Securities Australia Ltd Buy 1.7304 1.15.663 UBS Securities Australia Ltd Buy 5.731 5.397 UBS Securities Australia Ltd Buy 5.731 5.397 UBS Securities Australia Ltd Buy 5.731 5.307 UBS Securities Australia Ltd Buy 5.731 7.271 6.275 UBS Securities			29 23,395 37,010 40,992 29,395 29,395 20 112,904		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Constructive Australia Ltd Buy 25 3 USS Securities Australia Ltd Buy 239 3 UBS Securities Australia Ltd Buy 27,010 37,22 UBS Securities Australia Ltd Buy 23,395 2,397 UBS Securities Australia Ltd Buy 23,395 2,397 UBS Securities Australia Ltd Buy 20,395 4,200 UBS Securities Australia Ltd Buy 20,395 4,200 UBS Securities Australia Ltd Buy 2,284 982 UBS Securities Australia Ltd Buy 1,294 1,1568 UBS Securities Australia Ltd Buy 2,251 2,39 UBS Securities Australia Ltd Buy 1,734 4,276 UBS Securities Australia Ltd Buy 1,734 1,263 UBS Securities Australia Ltd Buy 1,734 1,265 UBS Securities Australia Ltd Buy 1,734 1,275 UBS Securities Australia Ltd Buy 5,281 5,937 UBS Securities Australia Ltd Buy			23,395 23,395 37,010 40,992 20,992 20 112,904 112,904		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Loss Securities Australia Ltd Buy 23,395 2,397 UBS Securities Australia Ltd Buy 23,395 2,397 UBS Securities Australia Ltd Buy 37,010 3,792 UBS Securities Australia Ltd Buy 23,395 2,397 UBS Securities Australia Ltd Buy 20 2 UBS Securities Australia Ltd Buy 5,738 5,936 UBS Securities Australia Ltd Buy 5,731 12,733 UBS Securities Australia Ltd Buy 5,731 5,397 UBS Securities Australia Ltd Buy			23,395 37,010 40,992 23,395 39 112,904 112,904		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 37,010 3.992 UBS Securities Australia Ltd Buy 37,010 3.792 UBS Securities Australia Ltd Buy 37,010 3.792 UBS Securities Australia Ltd Buy 23,395 2,397 UBS Securities Australia Ltd Buy 20 2 UBS Securities Australia Ltd Buy 5,938 2,396 UBS Securities Australia Ltd Buy 5,293 2,396 UBS Securities Australia Ltd Buy 5,293 2,396 UBS Securities Australia Ltd Buy 5,251 2,396 UBS Securities Australia Ltd Buy 5,251 3,184 UBS Securities Australia Ltd Buy 5,251 3,537 UBS Securities Australia Ltd Buy 5,251 3,637 UBS Securities Australia Ltd Buy 5,758 3,750 UBS Securities Australia Ltd Buy 5,758 3,750 UBS Securities Australia Ltd Buy 5,758 3,750 UBS Securities Australia Ltd		uny uny uny uny uny uny uny	23.395 37,010 40,905 23.395 20 112,904		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Bluy 37,010 37,92 UBS Securities Australia Ltd Bluy 40,992 4,209 UBS Securities Australia Ltd Bluy 23,395 4,209 UBS Securities Australia Ltd Bluy 20 2 UBS Securities Australia Ltd Bluy 20 2 UBS Securities Australia Ltd Bluy 57,939 5,936 UBS Securities Australia Ltd Bluy 57,939 20 UBS Securities Australia Ltd Bluy 57,939 5,936 UBS Securities Australia Ltd Bluy 1,156 1,164 UBS Securities Australia Ltd Bluy 1,291,671 1,164 UBS Securities Australia Ltd Bluy 1,7304 1,7304 UBS Securities Australia Ltd Bluy 1,7304 1,7304 UBS Securities Australia Ltd Bluy 1,7304 1,7304 UBS Securities Australia Ltd Bluy 5,716 5,900 UBS Securities Australia Ltd Bluy 5,716 5,716 UBS Securities Australia Ltd<	1 1 1 1 1 1 1 1 1 1	iny iny iny iny iny iny iny iny iny iny	37,010 40,992 23,395 39 39 112,904		Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 4200 UBS Securities Australia Ltd Buy 2395 4 200 UBS Securities Australia Ltd Buy 20 2 UBS Securities Australia Ltd Buy 7,935 5.96 UBS Securities Australia Ltd Buy 7,935 5.96 UBS Securities Australia Ltd Buy 2.0 2 UBS Securities Australia Ltd Buy 5.261 2 UBS Securities Australia Ltd Buy 2.0 2 UBS Securities Australia Ltd Buy 1,184 9.584 982 UBS Securities Australia Ltd Buy 1,7304 1,184 1.184 UBS Securities Australia Ltd Buy 1,7304 1,1263 1.184 UBS Securities Australia Ltd Buy 1,034,712 1,1263 UBS		uy buy buy buy buy	40,992 23,395 39 20 112,904	2,397 2,397 4 4 2 2	Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 23,395 2,397 UBS Securities Australia Ltd Buy 112,904 11,568 UBS Securities Australia Ltd Buy 112,904 11,568 UBS Securities Australia Ltd Buy 5,935 5,936 UBS Securities Australia Ltd Buy 5,7,935 5,936 UBS Securities Australia Ltd Buy 5,261 2,0 UBS Securities Australia Ltd Buy 5,251 5,38 UBS Securities Australia Ltd Buy 1,173 20 UBS Securities Australia Ltd Buy 1,173 1,173 UBS Securities Australia Ltd Buy 1,173 1,173 UBS Securities Australia Ltd Buy 1,730 1,173 UBS Securities Australia Ltd Buy 1,730 1,175 UBS Securities Australia Ltd Buy 1,725 1,269 UBS Securities Australia Ltd Buy 1,727 1,126 UBS Securities Australia Ltd Buy 5,169 5,100 UBS Securities Australia Ltd		iny iny iny iny	23,395 39 20 112,904	2,397 4 2 2 11,568	Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 39 4 UBS Securities Australia Ltd Buy 112,904 11,586 UBS Securities Australia Ltd Buy 5,935 5,936 UBS Securities Australia Ltd Buy 5,035 5,936 UBS Securities Australia Ltd Buy 5,037 20 UBS Securities Australia Ltd Buy 5,251 5,88 UBS Securities Australia Ltd Buy 1,091,671 11,263 UBS Securities Australia Ltd Buy 1,091,671 112,663 UBS Securities Australia Ltd Buy 1,091,671 12,009 <td< td=""><td>1 </td><td>\(\frac{1}{2}\) \(\frac{1}{2}\) \(\frac{1}2\) \(\frac{1}{2}\) \(\frac{1}2\) \(\frac{1}2\) \(\frac{1}2\) \(\frac{1}2\) \(\frac\</td><td>39 20 20 112,904</td><td>4 2 2 11,568</td><td>Ordinary Ordinary Ordinary</td></td<>	1	\(\frac{1}{2}\) \(\frac{1}2\) \(\frac{1}{2}\) \(\frac{1}2\) \(\frac{1}2\) \(\frac{1}2\) \(\frac{1}2\) \(\frac\	39 20 20 112,904	4 2 2 11,568	Ordinary Ordinary Ordinary
UBS Securities Australia Ltd Buy 10.204 11.588 UBS Securities Australia Ltd Buy 10.204 11.588 UBS Securities Australia Ltd Buy 5.7,835 5.986 UBS Securities Australia Ltd Buy 5.7,835 5.986 UBS Securities Australia Ltd Buy 5.7,831 5.88 UBS Securities Australia Ltd Buy 5.251 5.88 UBS Securities Australia Ltd Buy 5.251 5.88 UBS Securities Australia Ltd Buy 5.251 5.88 UBS Securities Australia Ltd Buy 5.7,811 5.937 UBS Securities Australia Ltd Buy 5.7,811 5.937 UBS Securities Australia Ltd Buy 5.7,811 5.937 UBS Securities Australia Ltd Buy 5.7,811 6.367 UBS Securities Australia Ltd Buy 5.7,811 6.346 UBS Securities Australia Ltd Buy 5.7,812 6.346 UBS Securities Australia Ltd Buy 75,799 6.346 UBS Securities Australia	1 1 1 1 1 1 1 1 1 1 1 1	LUY LUY LUY	112,904	11,568	Ordinary Ordinary
UBS Securities Australia Ltd Buy 112,904 1,1588 UBS Securities Australia Ltd Buy 57,935 5,936 UBS Securities Australia Ltd Buy 57,935 5,936 UBS Securities Australia Ltd Buy 20 2 UBS Securities Australia Ltd Buy 1,1556 1,184 UBS Securities Australia Ltd Buy 1,526 1,184 UBS Securities Australia Ltd Buy 1,1556 1,184 UBS Securities Australia Ltd Buy 1,156 1,173 UBS Securities Australia Ltd Buy 1,091,671 1,12,633 UBS Securities Australia Ltd Buy 57,169 4,2346) UBS Securities Australia Ltd Buy 57,169 1,200 UBS	1 1 1 1 1 1 1	iuy iuy	112,904	11,568	Ordinary
UBS Securities Australia Ltd Buy 11,2904 11,588 UBS Securities Australia Ltd Buy 57,935 5,936 UBS Securities Australia Ltd Buy 5,936 22 UBS Securities Australia Ltd Buy 20 2 UBS Securities Australia Ltd Buy 5,584 982 UBS Securities Australia Ltd Buy 1,7304 1,773 UBS Securities Australia Ltd Buy 1,556 1,184 UBS Securities Australia Ltd Buy 1,5304 1,773 UBS Securities Australia Ltd Buy 1,594,712 11,5663 UBS Securities Australia Ltd Buy 5,581 4,276 UBS Securities Australia Ltd Buy 1,691,671 11,5693 UBS Securities Australia Ltd Buy 1,691,771 11,5693 UBS Securities Australia Ltd Buy 5,781 4,276 UBS Securities Australia Ltd Buy 5,773 4,276 UBS Securities Australia Ltd Buy 5,773 7,720 UBS Securities Australia Ltd <td>11111111</td> <td>iuy</td> <td>112,904</td> <td>11,568</td> <td>Ordinary</td>	11111111	iuy	112,904	11,568	Ordinary
UBS Securities Australia Ltd Buy 57,936 UBS Securities Australia Ltd Buy 57,935 5,936 UBS Securities Australia Ltd Buy 20 2 UBS Securities Australia Ltd Buy 1,184 20 UBS Securities Australia Ltd Buy 1,1556 1,184 UBS Securities Australia Ltd Buy 1,7304 1,773 UBS Securities Australia Ltd Buy 1,7304 1,773 UBS Securities Australia Ltd Buy 1,7304 1,773 UBS Securities Australia Ltd Buy 1,091,671 (112,663) UBS Securities Australia Ltd Buy 1,691,671 (112,663) UBS Securities Australia Ltd Buy 4,425 6,337 UBS Securities Australia Ltd Buy 5,7169 12,690 UBS Securities Australia Ltd Buy 5,7169 12,730 UBS Securities Australia Ltd Buy 5,7169 12,730 UBS Securities Australia Ltd Buy 5,7169 12,730 UBS Securities Australia Ltd	11111111	á í	20		
UBS Securities Australia Ltd Buy 5,935 5,936 UBS Securities Australia Ltd Buy 5,034 922 UBS Securities Australia Ltd Buy 20 2 UBS Securities Australia Ltd Buy 20 2 UBS Securities Australia Ltd Buy 1,7304 1,7304 UBS Securities Australia Ltd Buy 1,7304 1,7305 UBS Securities Australia Ltd Buy 1,091,671 1,12,663 UBS Securities Australia Ltd Buy 1,091,671 1,12,663 UBS Securities Australia Ltd Buy 1,091,671 1,12,663 UBS Securities Australia Ltd Buy 1,16,805 1,206 UBS Securities Australia Ltd Buy 1,16,805 1,206 UBS Securities Australia Ltd Buy 1,16,805 1,200 UBS Securities Australia Ltd Buy 1,16,805 1,200 UBS Securities Australia Ltd Buy 1,10,91,671 1,12,603 UBS Securities Australia Ltd Buy 2,109 (4,306) UBS Securit		22	- 1	3	Ordinary
UBS Securities Australia Ltd Buy 20 UBS Securities Australia Ltd Buy 9,584 962 UBS Securities Australia Ltd Buy 1,1536 1,184 UBS Securities Australia Ltd Buy 1,2326 1,184 UBS Securities Australia Ltd Buy 1,2326 1,184 UBS Securities Australia Ltd Buy 1,291,671 1,773 UBS Securities Australia Ltd Buy 1,594,172 1,12633 UBS Securities Australia Ltd Buy 1,094,172 1,12633 UBS Securities Australia Ltd Buy 1,094,173 1,12633 UBS Securities Australia Ltd Buy 5,169 5,300 UBS Securities Australia Ltd Buy 5,1,109 1,346) UBS Securities Australia Ltd Buy 75,179 1,200 UBS Securities Australia		·	o	2,936	Ordinary
UBS Securities Australia Ltd Buy 9.584 9.82 UBS Securities Australia Ltd Buy 1.556 1,184 UBS Securities Australia Ltd Buy 1.556 1,184 UBS Securities Australia Ltd Buy 1.091,671 1.1263 UBS Securities Australia Ltd Buy 1.091,671 1.12,663 UBS Securities Australia Ltd Buy 5,581 5,937 UBS Securities Australia Ltd Buy 57,581 6,937 UBS Securities Australia Ltd Buy 1,091,671 112,663 UBS Securities Australia Ltd Buy 1,180 1,069 UBS Securities Australia Ltd Buy 1,16,07 1,12,633 UBS Securities Australia Ltd Buy 1,16,07 1,12,633 UBS Securities Australia Ltd Buy 1,10,091,671 112,663 UBS Securities Australia Ltd Buy 5,109 (4,346) UBS Securities Australia Ltd Sell 42,109 (4,346) UBS Securities Australia Ltd Buy 3,078 3,078		À	20	2	Ordinary
UBS Securities Australia Ltd Buy 9,584 982 UBS Securities Australia Ltd Buy 1,154 2 UBS Securities Australia Ltd Buy 1,256 1,184 UBS Securities Australia Ltd Buy 1,2304 1,773 UBS Securities Australia Ltd Buy 1,7304 1,773 UBS AC London Branch Sell 1,091,671 112,663 UBS AC London Branch Sell 1,094,772 112,663 UBS AC London Branch Sell 1,094,772 112,663 UBS Securities Australia Ltd Buy 41,425 4,276 UBS Securities Australia Ltd Buy 1,091,671 116,579 UBS Securities Australia Ltd Buy 57,769 5,300 UBS Securities Australia Ltd Buy 57,799 5,300 UBS Securities Australia Ltd Buy 75,799 5,300 UBS Securities Australia Ltd Buy 75,799 7,720 UBS Securities Australia Ltd Buy 75,799 7,720 UBS Securities Australia Ltd	11111	NII.	30	2	Crding
UBS Securities Australia Ltd Buy 1,184 UBS Securities Australia Ltd Buy 1,185 1,184 UBS Securities Australia Ltd Buy 1,256 1,184 UBS Securities Australia Ltd Buy 1,091,671 (11,2663) UBS Securities Australia Ltd Buy 1,091,671 (11,2663) UBS Securities Australia Ltd Buy 5,781 36,733 UBS Securities Australia Ltd Buy 41,425 4,276 UBS Securities Australia Ltd Buy 41,425 4,276 UBS Securities Australia Ltd Buy 5,937 4,276 UBS Securities Australia Ltd Buy 5,7169 5,900 UBS Securities Australia Ltd Buy 5,7169 5,000 UBS Securities Australia Ltd Buy 5,7169 5,000 UBS Securities Australia Ltd Buy 5,7169 4,346 UBS Securities Australia Ltd Buy 5,7169 4,346 UBS Securities Australia Ltd Buy 2,221 (5,29) UBS Securities Australia Ltd	1 1	N.I.	9 584	080	dedica
USS Securities Australia Ltd Buy 11,556 1,184 UBS Securities Australia Ltd Buy 1,251 5.88 UBS Securities Australia Ltd Buy 1,094,712 1,263 UBS Securities Australia Ltd Buy 1,094,712 1,263 UBS Securities Australia Ltd Buy 5,781 36,126 UBS Securities Australia Ltd Buy 5,781 4,276 UBS Securities Australia Ltd Buy 1,16,077 1,10,663 UBS Securities Australia Ltd Buy 1,10,677 1,10,663 UBS Securities Australia Ltd Sell 42,109 (5,000) UBS Securities Australia Ltd Buy 3,01,671 2,271 UBS Securities Australia Ltd Buy 3,078 3,078 UBS Securities Australia Ltd Buy 3,078 3,078 <	111	NII.	O.C.	3	in in it
UBS Securities Australia Ltd Buy 1,784 UBS Securities Australia Ltd Buy 1,730 1,733 UBS Securities Australia Ltd Buy 1,034,172 1,12663 UBS Securities Australia Ltd Buy 1,034,172 (112,663) UBS Securities Australia Ltd Buy 5,781 5,937 UBS Securities Australia Ltd Buy 4,425 4,276 UBS Securities Australia Ltd Buy 1,094,717 (16,575) UBS Securities Australia Ltd Buy 1,094,717 (16,576) UBS Securities Australia Ltd Buy 1,094,717 (16,576) UBS Securities Australia Ltd Buy 5,169 5,300 UBS Securities Australia Ltd Buy 5,169 5,300 UBS Securities Australia Ltd Sell 3,178 3,1567 UBS Securities Australia Ltd Buy 75,799 7,720 UBS Securities Australia Ltd Buy 75,799 7,720 UBS Securities Australia Ltd Buy 3,169 3,16 UBS Securities Australia Ltd	111	, n	07		Gruinary
UBS Securities Australia Ltd Buy 1,091,671 538 UBS Securities Australia Ltd Buy 1,091,671 1,12,633 UBS AC London Branch Buy 1,091,671 112,663 UBS Securities Australia Ltd Buy 5,581 5,937 UBS Securities Australia Ltd Buy 57,581 5,937 UBS Securities Australia Ltd Buy 1,091,671 112,663 UBS Securities Australia Ltd Buy 1,081,671 112,663 UBS Securities Australia Ltd Buy 5,179 4,270 UBS Securities Australia Ltd Buy 3,01,78 3,07 UBS Securities Australia Ltd Buy 3,07 3,07 UBS Securities Australia Ltd Buy 3,07 3,07 UBS Securities Australia Ltd Buy 3,07 3,07 UBS Securi	11	ń	955,11		Ordinary
UBS Securities Australia Ltd Buy 17.304 1.773 UBS AG London Branch Sell 1,094,712 112,663 UBS AG London Branch Sell 1,094,712 (112,663) UBS Securities Australia Ltd Buy 356,126 36,733 UBS Securities Australia Ltd Buy 41,425 4,276 UBS Securities Australia Ltd Buy 1,091,671 11,663 UBS Securities Australia Ltd Buy 57,169 5,900 UBS Securities Australia Ltd Sell 75,783 37,657 UBS Securities Australia Ltd Buy 75,793 38 UBS Securities Australia Ltd Buy 75,793 48 50 UBS Securities Australia Ltd Buy 2,221 72,29 48 UBS Securities Australia Ltd Buy 3,078 318	- 1	ń	5,251	238	Ordinary
UBS AGC Lundon Branch Bby 1,091,671 112,663 UBS AG Lundon Branch Sell 1,091,671 (112,663) UBS AG Lundon Branch Bby 356,126 36,733 UBS Securities Australia Ltd Buy 4,775 4,276 UBS Securities Australia Ltd Buy 4,742 4,276 UBS Securities Australia Ltd Buy 1,691,671 1,0563 UBS Securities Australia Ltd Buy 1,091,671 1,0563 UBS Securities Australia Ltd Buy 5,7169 4,346) UBS Securities Australia Ltd Sell 45,109 (4,346) UBS Securities Australia Ltd Sell 36,783 (5,000) UBS Securities Australia Ltd Buy 3,7169 (4,346) UBS Securities Australia Ltd Buy 3,783 3,78 UBS Securities Australia Ltd Buy 3,783 3,78 UBS Securities Australia Ltd Buy 3,78 3,78 UBS Securities Australia Ltd Buy 4,65 3,78 UBS Securities Australia Ltd <td></td> <td>iny</td> <td>17,304</td> <td>1,773</td> <td>Ordinary</td>		iny	17,304	1,773	Ordinary
UBS AGLORIdon Branch Sell 1,094,712 (112,663) UBS Securities Australia Ltd Buy 35,753 36,733 UBS Securities Australia Ltd Buy 41,425 4,276 UBS Securities Australia Ltd Buy 116,0771 11,663 UBS Securities Australia Ltd Buy 1,091,677 11,663 UBS Securities Australia Ltd Buy 1,091,677 11,663 UBS Securities Australia Ltd Buy 1,091,677 11,663 UBS Securities Australia Ltd Buy 25,169 5,500 UBS Securities Australia Ltd Sell 42,109 (4,346) UBS Securities Australia Ltd Sell 36,78 37,670 UBS Securities Australia Ltd Sell 2,21,79 (5,000) UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 3,078 38 UBS Securities Australia Ltd Buy 3,078 3,078 UBS Securities Australia Ltd Buy 5,391 48 UBS Securities A		ÁΩ	1,091,671		Ordinary
UBS Securities Australia Ltd Buy 356,126 38,733 UBS Securities Australia Ltd Buy 57,581 5,937 UBS Securities Australia Ltd Buy 41,426 4,226 UBS Securities Australia Ltd Buy 116,805 11,069 UBS Securities Australia Ltd Buy 5,7,69 5,900 UBS Securities Australia Ltd Buy 5,7,69 5,000 UBS Securities Australia Ltd Sell 42,109 (4,346) UBS Securities Australia Ltd Sell 364,783 (5,000) UBS Securities Australia Ltd Sell 48,300 (5,000) UBS Securities Australia Ltd Buy 3,221 (5,000) UBS Securities Australia Ltd Buy 3,078 3,18 UBS Securities Australia Ltd Buy 2,221 (5,000) UBS Securities Australia Ltd Buy 2,291 48 UBS Securities Australia Ltd Buy 5,900 48 UBS Securities Australia Ltd Buy 5,391 5,71 UBS Securities Australia	ı	[]	1 094 717		Ordinary
UBS Securities Australia Ltd Buy 57,381 5,781 UBS Securities Australia Ltd Buy 57,381 5,787 UBS Securities Australia Ltd Buy 116,807 12,069 UBS Securities Australia Ltd Buy 116,807 12,069 UBS Securities Australia Ltd Buy 57,169 5,900 UBS Securities Australia Ltd Buy 57,169 5,900 UBS Securities Australia Ltd Buy 57,169 7,570 UBS Securities Australia Ltd Sell 42,109 (5,000) UBS Securities Australia Ltd Sell 3,078 3,078 UBS Securities Australia Ltd Buy 4,65 48 UBS Securities Australia Ltd Buy 6,590 6,590 UBS Securities Australia Ltd Buy 6,590 6,605 UBS Securities Australia Ltd <td>ı</td> <td>Ail</td> <td>356 126</td> <td>36 753</td> <td>idenipo</td>	ı	Ail	356 126	36 753	idenipo
USS Securities Australia Ltd Buy 41,425 4,276 UBS Securities Australia Ltd Buy 11,6805 12,069 UBS Securities Australia Ltd Buy 1,091,671 11,563 UBS Securities Australia Ltd Buy 5,169 5,900 UBS Securities Australia Ltd Buy 5,169 5,900 UBS Securities Australia Ltd Buy 75,169 5,900 UBS Securities Australia Ltd Sell 75,169 5,000 UBS Securities Australia Ltd Sell 364,783 (5,000) UBS Securities Australia Ltd Buy 364,783 (5,000) UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 2,221 (5,000) UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 5,391 48 UBS Securities Australia Ltd Buy 5,392 557 UBS Securities Australia Ltd Buy 5,392 557 UBS Securities Australia Ltd <td>T</td> <td>(a)</td> <td>121,020</td> <td>20,700</td> <td>Cidinaly</td>	T	(a)	121,020	20,700	Cidinaly
UBS Securities Australia Ltd Self 1,725 4,275 UBS Securities Australia Ltd Buy 1,607,71 11,069 UBS Securities Australia Ltd Buy 1,091,671 12,069 UBS Securities Australia Ltd Buy 7,169 5,300 UBS Securities Australia Ltd Buy 7,169 5,300 UBS Securities Australia Ltd Buy 7,179 4,340 UBS Securities Australia Ltd Sell 42,109 (4,340) UBS Securities Australia Ltd Sell 3,221 (2,29) UBS Securities Australia Ltd Buy 3,078 3,078 UBS Securities Australia Ltd Buy 3,078 3,078 UBS Securities Australia Ltd Buy 3,078 3,078 UBS Securities Australia Ltd Buy 465 48 UBS Securities Australia Ltd Buy 5,391 465 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd	Ť	â	100,70	156'6	Ordinary
UBS Securities Australia Ltd Seli 16,0771 16,0751 UBS Securities Australia Ltd Buy 1,087 1,063 UBS Securities Australia Ltd Buy 1,081,677 1,12,633 UBS Securities Australia Ltd Buy 5,109 5,900 UBS Securities Australia Ltd Buy 751,793 7,757 UBS Securities Australia Ltd Sell 48,300 (5,000) UBS Securities Australia Ltd Sell 36,783 (37,657) UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 3,078 3,078 UBS Securities Australia Ltd Buy 5,391 2,47 UBS Securities Australia Ltd Buy 5,391 2,47 UBS Securities Australia Ltd Buy 5,391 3,57 UBS Securities Australia Ltd Buy 5,392 5,57 UBS Securities Australia Ltd </td <td>7</td> <td>Ś</td> <td>41,425</td> <td>4,276</td> <td>Ordinary</td>	7	Ś	41,425	4,276	Ordinary
UBS Securities Australia Ltd Buy 116,805 12,069 UBS Securities Australia Ltd Buy 1,091,671 112,663 UBS Securities Australia Ltd Buy 5,7169 5,900 UBS Securities Australia Ltd Buy 5,7169 (4,346) UBS Securities Australia Ltd Sell 36,783 77,570 UBS Securities Australia Ltd Buy 36,783 77,570 UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 2,391 247 UBS Securities Australia Ltd Buy 465 48 UBS Securities Australia Ltd Buy 6,391 57 UBS Securities Australia Ltd Buy 6,393 57 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Bu	١	ell	160,771	(16,575)	Ordinary
UBS Securities Australia Ltd Buy 1,091,671 112,663 UBS Securities Australia Ltd Sell 4,300 (4,306) UBS Securities Australia Ltd Buy 751,793 77,570 UBS Securities Australia Ltd Sell 751,793 77,570 UBS Securities Australia Ltd Sell 751,793 77,570 UBS Securities Australia Ltd Sell 36,783 77,570 UBS Securities Australia Ltd Buy 484 50 UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 5,392 48 UBS Securities Australia Ltd Buy 5,392 48 UBS Securities Australia Ltd Buy 5,392 557 UBS Securities Australia Ltd Buy 5,392 577 UBS Securities Australia Ltd Buy 6,396 661 UBS Securities Australia Ltd Buy 6,396 661 UBS Securities Australia Ltd <		'n	116,805	12,069	Ordinary
UBS Securities Australia Ltd Buy 57,169 5,900 UBS Securities Australia Ltd Sell 42,109 (4,346) UBS Securities Australia Ltd Sell 75,793 77,500 UBS Securities Australia Ltd Sell 364,783 (7,500) UBS Securities Australia Ltd Buy 48,300 (5,000) UBS Securities Australia Ltd Buy 36,78 318 UBS Securities Australia Ltd Buy 2,391 378 UBS Securities Australia Ltd Buy 3,156 376 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 5,392 557 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd B		À	1.091.671	112.663	Ordinary
UBS Securities Australia Ltd Sell 42,109 (4,246) UBS Securities Australia Ltd Buy 75,70 (3,500) UBS Securities Australia Ltd Sell 364,783 (5,000) UBS Securities Australia Ltd Sell 3,221 (229) UBS Securities Australia Ltd Buy 3,078 3,078 UBS Securities Australia Ltd Buy 3,078 347 UBS Securities Australia Ltd Buy 3,078 347 UBS Securities Australia Ltd Buy 465 48 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 6,392 557 UBS Securities Australia Ltd Buy 6,392 661 UBS Securities Australia Ltd Buy 6,392 661 UBS Securities Australia Ltd Buy 4,840 1,50 UBS Securities Australia Ltd Buy 4,840 1,50 UBS Securities Australia Ltd Buy 1,510 1,50		All	57 169	2 900	Ordinan
UBS Securities Australia Ltd Buy 751,793 77,70 UBS Securities Australia Ltd Sell 48,300 (5,000) UBS Securities Australia Ltd Sell 36,783 (7,570 UBS Securities Australia Ltd Buy 484 50 UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 2,391 2,47 UBS Securities Australia Ltd Buy 3,156 3,57 UBS Securities Australia Ltd Buy 5,392 5,57 UBS Securities Australia Ltd Buy 5,392 5,57 UBS Securities Australia Ltd Buy 5,392 5,57 UBS Securities Australia Ltd Buy 6,396 661 UBS Securities Australia Ltd Buy 6,396 661 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 1,510 156	l	10	42 100	(975 7)	Calibas
USS Securities Australia Ltd Sell 7,370 USS Securities Australia Ltd Sell 36,783 (7,500) USS Securities Australia Ltd Sell 2,221 (5,200) UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 2,391 347 UBS Securities Australia Ltd Buy 2,391 347 UBS Securities Australia Ltd Buy 465 348 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 4,840 50 UBS Securities Australia Ltd Buy 1,510 1,510	ı		201,200	01011	1000
USS Securities Australia Ltd Sell 364,789 (5,000) USS Securities Australia Ltd Sell 2,221 (7,229) UBS Securities Australia Ltd Buy 484 50 UBS Securities Australia Ltd Buy 3,078 3,18 UBS Securities Australia Ltd Buy 2,391 247 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 6,392 48 UBS Securities Australia Ltd Buy 6,392 557 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 1,220 126 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 1,510 1,56	ı	à T	267,167	0/5//	Ordinary
UBS Securities Australia Ltd Sell 364,783 (37,657) UBS Securities Australia Ltd Sull 484 50 UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 2,391 247 UBS Securities Australia Ltd Buy 2,391 3,26 UBS Securities Australia Ltd Buy 465 48 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 6,396 661 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 4,840 106 UBS Securities Australia Ltd Buy 4,840 106 UBS Securities Australia Ltd Buy 1,510 156	ı		46,300	(nnn's)	Ordinary
UBS Securities Australia Ltd Skil 2,221 (229) UBS Securities Australia Ltd Buy 48,44 50 UBS Securities Australia Ltd Buy 3,078 316 UBS Securities Australia Ltd Buy 3,591 247 UBS Securities Australia Ltd Buy 465 36 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 6,392 651 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 1,220 126 UBS Securities Australia Ltd Buy 4,840 1,50 UBS Securities Australia Ltd Buy 1,510 1,50	1	lia li	364,783	(/59'/5)	Ordinary
UBS Securities Australia Ltd Buy 484 50 UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 2,391 247 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 6,805 773 UBS Securities Australia Ltd Buy 6,392 557 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 4,840 126 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 1,510 156	٦	e	2,221	(229)	Ordinary
UBS Securities Australia Ltd Buy 3,078 318 UBS Securities Australia Ltd Buy 2,391 247 UBS Securities Australia Ltd Buy 465 48 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 6,392 557 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 4,840 156 UBS Securities Australia Ltd Buy 1,510 156		σy	484	20	Ordinary
UBS Securities Australia Ltd Buy 2,391 247 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 465 48 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 5,392 557 UBS Securities Australia Ltd Buy 1,220 166 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 1,510 156		'n	3,078	318	Ordinary
UBS Securities Australia Ltd Buy 3,156 3.26 UBS Securities Australia Ltd Buy 465 48 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 5,392 557 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 4,840 1,50 UBS Securities Australia Ltd Buy 1,510 156	ŀ	A1	2 391	747	Ordinary
UBS Securities Australia Ltd Buy 5,392 37.0 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 5,392 557 UBS Securities Australia Ltd Buy 1,220 126 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 1,510 156	l		3 156	328	Coloro,
UBS Securities Australia Ltd Buy 6.805 78 UBS Securities Australia Ltd Buy 6.805 703 UBS Securities Australia Ltd Buy 6.398 661 UBS Securities Australia Ltd Buy 1,220 126 UBS Securities Australia Ltd Buy 4.840 500 UBS Securities Australia Ltd Buy 1,510 156	ı	5	00,10	075) (d
UBS Securities Australia Ltd Buy 6,805 703 UBS Securities Australia Ltd Buy 5,392 5,57 UBS Securities Australia Ltd Buy 6,398 661 UBS Securities Australia Ltd Buy 1,220 1,56 UBS Securities Australia Ltd Buy 4,840 1,50 UBS Securities Australia Ltd Buy 1,510 1,56	ı	ń	465	48	Ordinary
UBS Securities Australia Ltd Buy 5,392 557 UBS Securities Australia Ltd Buy 6,298 661 UBS Securities Australia Ltd Buy 1,220 1,26 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 1,510 156	Ť	ń	c08'q	703	Ordinary
UBS Securities Australia Ltd Buy 6.398 661 UBS Securities Australia Ltd Buy 1,220 126 UBS Securities Australia Ltd Buy 4,840 50 UBS Securities Australia Ltd Buy 1,510 156	_	'n	5,392	257	Ordinary
UBS Securities Australia Ltd Buy 1,220 176 UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 1,510 156	_	'n	86£'9	199	Ordinary
UBS Securities Australia Ltd Buy 4,840 500 UBS Securities Australia Ltd Buy 1,510 156		3	1,220	126	Ordinary
UBS Securities Australia Ltd Buy 1,510 156	Ī	AI.	4.840	200	Ordinary
UDS Securities Australia Ltd Buy 156	Τ		013.	200	2 15 15
	T	uý.	ULS,I	951	Ordinary

04-Jul-14 UBS Sec. 04-Jul-14 UBS	UBS Securities Australia Ltd UBS Securities Australia Ltd		68 68 68 68 68 68 68 68 703 704 704 704 68 68 68 68 68 68	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	Ordinary Ordinary
	aurities Australie Ltd aurities Australie Ltd		68 68 68 68 68 68 68 68 68 68 68 68 68 6	(10) (10) (10) (10) (10) (10) (10) (10)	Ordinary Ordinary
	untites Australia Ltd untites Australia Ltd		68 68 68 68 68 68 68 68 68 68 68 68 68 6	(7) (7) (7) (7) (7) (7) (7) (7) (7) (7)	Ordinary Ordinary
	untites Australia Ltd untites Australia Ltd		68 68 68 68 68 68 68 68 68 68 68 68 68 6	(100 (100 (100 (100 (100 (100 (100 (100	Ordinary Ordinary
	unties Australia Ltd unties Australia Ltd		68 68 68 68 68 68 68 70 70 70 70 70 70 70 70 70 70 70 70 70	(7) (7) (7) (7) (7) (7) (7) (7) (7) (7)	Ordinary Ordinary
	untites Australia Ltd untites Australia Ltd		68 68 68 68 68 68 68 704 704 704 704 704 704 68 68 68 68 68	(7) (7) (7) (7) (7) (7) (7) (7) (7) (7)	Odinary Ordinary
	untrites Australia Ltd untrites Australia Ltd		68 68 68 68 68 68 703 704 704 703 703 704 68 68 68	(7) (7) (7) (7) (7) (10) (7) (10) (10) (10) (10) (10) (10) (10) (10	Ordinary Ordinary
	ururites Australia Ltd ururites Australia Ltd		68 68 68 68 68 68 704 704 705 705 705 705 705 705 705 705 705 705	(7) (7) (7) (7) (105) (1	Ordinary Ordinary
	untrites Australia Ltd untrites Australia Ltd		68 68 68 68 68 704 704 704 704 68 68 68 68 68 68	(7) (7) (7) (7) (1058)	Ordinary Ordinary
	untries Australia Ltd untries Australia Ltd		68 68 68 68 68 704 68 68 68 68 68 68	(72) (73) (73) (73) (73) (73) (74) (74) (74) (74) (74) (74) (74) (74	Ordinary Ordinary
	untites Australia Ltd untites Australia Ltd		68 68 68 703 703 703 703 703 703 703 703 703 703	(7) (7) (7) (105)	Ordinary Ordinary
	unties Australia Ltd unties Australia Ltd		68 68 704 704 704 1,027 1,027 2,641 68 68 68 68 68 68	(77) (78) (105) (105) (105) (105) (107) (1	Ordinary Ordinary
	untites Australie Ltd untites Australie Ltd		704 68 68 68 68 68 68 68 68 68 68 68 68 68	(72) (105) (Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	unities Australia Ltd unities Australia Ltd		704 68 763 1,027 2,641 68 4,049 68 4,538 68 68	(72) (78) (78) (70) (70) (70) (70) (70) (70) (70) (70	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	turnites Australia Ltd unities Australia Ltd		763 763 1,027 2,641 68 68 4,049 68 68 68 68	(7) (105) (105) (105) (105) (105) (107) (1	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	unities Australia Ltd unities Australia Ltd		763 1,027 2,641 68 68 68 68 4,538 68 68	(105) (105) (105) (107)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	untites Australia Ltd untites Australia Ltd		1,027 2,641 68 68 4,049 68 4,538 4,538 68 68	(105) (77) (71 (414) (71 (70) (71 (70) (70) (70) (70) (70) (70)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1111	unrites Australia Ltd unrites Australia Ltd		2.641 6.8 6.8 6.9 6.8 6.8 6.8 6.8 6.8 6.8	(270) (7) (7) (7) (7) (7) (7) (7) (7) (7) (7	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	unties Austrella Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd		68 68 68 68 68 68 68 68 68	(7) (7) (7) (7) (7) (7) (7) (7) (7) (7)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd untites Australia Ltd		4,049 68 68 68 68 68 68	(41.4) (7) (7) (7) (7) (7) (7) (7) (7) (7) (7	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd		4,049 68 68 68 68 68 68	(414) (7) (7) (7) (7) (7) (7) (7) (7) (7) (7	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Т	unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd	3, 3, 3, 3, 3, 3, 3, 3, 3, 3, 3, 3, 3, 3	68 4,538 68 68	(7) (464) (7) (7) (7) (7) (7) (7) (7) (7) (7) (7	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	unities Australia Ltd unties Australia Ltd	3. 5. 5. 5. 5. 5. 5. 5. 5. 5. 5. 5. 5. 5.	4,538 68 68	(7) (7) (7) (7) (7) (7) (2,021)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
ľ	urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	· · · · · · · · · · · · · · · · · · ·	4,538 68 68 68	(7) (7) (7) (7) (2,021)	Ordinary Ordinary Ordinary Ordinary Ordinary
-	urities Australia Lid urities Australia Lid urities Australia Lid urities Australia Lid urities Australia Lid urities Australia Lid urities Australia Lid urities Australia Lid	· · · · · · · · · · · · · · · · · · ·	89	(7) (7) (7) (2,021) (2,029)	Ordinary Ordinary Ordinary Ordinary Ordinary
-	unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd unties Australia Ltd	Sell Sell	89 99	(7) (7) (2,021)	Ordinary Ordinary Ordinary Ordinary
-	urites Australia Ltd urites Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Sell Sell	89	(7) (2,021) (2,029)	Ordinary Ordinary Ordinary
- 1	urites Australia Ltd urities Australia Ltd urities Australia Ltd urities Australia Ltd	Sell	. 646 4.	(2,021)	Ordinary Ordinary Ordinary
04-Jul-14 UBS Secu	urites Australia Lid urities Australia Lid urities Australia Lid	Sell	19,/65	(2,029)	Ordinary
- 1	urities Australia Ltd urities Australia Ltd		19,844		Ordinary
- 1	urities Australia Ltd	Sell	245	(25)	Ç
		Sell	89	(2)	Ordinary
٦	UBS Securities Australia Ltd	Sell	21,966	(2,246)	Ordinary
	UBS Securities Australia Ltd	Sell	29,086	(2,974)	Ordinary
	urities Australia Ltd	Sell	29	(3)	Ordinary
	UBS Securities Australia Ltd	Sell	89	(2)	Ordinary
	JBS Securities Australia Ltd	Sell	1,056	(108)	Ordinary
	UBS Securities Australia Ltd	Seli	3,149	(322)	Ordinary
	urities Australia Ltd	Sell	89	(2)	Ordinary
l	urities Australia Ltd	Seli	775	(23)	Ordinary
04-Jul-14 UBS Secu	UBS Securities Australia Ltd	Sell	89	(2)	Ordinary
ı	urities Australia Ltd		89	(2)	Ordinary
ı	urities Australia Ltd	198	89	(2)	Ordinary
L	unities Australia I to	و	89	3	Ordinan
	IJRS Securities Australia I to	وا	89	6	Ordinary
ı	Securities Australia Ltd	100	00		Ordinaly
T	Ullies Australia Ltd	lake in	80		Ordinary
7	UBS Securities Australia Ltd	Seli	4,479	(458)	Ordinary
٦	UBS Securities Australia Ltd	lleS.	89	(2)	Ordinary
٦	UBS Securities Australia Ltd	Sell	89	(7)	Ordinary
	UBS Securities Australia Ltd	Sell	89	(2)	Ordinary
	UBS Securities Australia Ltd	Sell	5,408	(553)	Ordinary
04-Jul-14 UBS Secu	urities Australia Ltd	Sell	23,276	(2,380)	Ordinary
1	urities Australia Ltd	Sell	5.154	(527)	Ordinary
ı	UBS Securities Australia Ltd	Seil	714	(73)	Ordinary
T	IBS Securities Australia I to	7	8 127	(831)	Ordinan
ı	urities Australia I to	Sell	147	(15)	Ordinany
	LIBC Countries Australia 14d	Soli	L 060	(003)	Ordinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	divisor Australia Lida	1100	2,000	(000)	Countries
1	Untres Australia Lto	llac I	53,634	(5,484)	Ordinary
	OBS Securities Australia Ltd	i i	7/5'77	(2,308)	Ordinary
	UBS Securities Australia Ltd	Sell	33,937	(3,470)	Ordinary
	UBS Securities Australia Ltd	Sell	705	(72)	Ordinary
	urities Australia Ltd	Seil	52,849	(5,401)	Ordinary
04-Jul-14 UBS Secur	UBS Securities Australia Ltd	Sell	18,034	(1,844)	Ordinary
Г	UBS Securities Australia Ltd	Sell	21,751	(2,224)	Ordinary
Γ	UBS Securities Australia Ltd	Sell	1 780	(182)	Ordinary
T	urition Australia 140	001	2 150	(000)	Ordinan
T	Obs securities Australia Ltd	liac -	3,139	(323)	Ordinary

08-Jul-14 UBS Securities Australia Ltd OB-Jul-14 UBS Secu	Mustralia Lid Australia Lid	Buy	8,199	847	Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	Buy	2010	416	Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd		4.027	-	
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	200	11,122	1149	Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	Buv	545 6	242	Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	Buy	4.617	477	Ordinary
	Australia Ltd Australia Ltd Australia Ltd	Buy	39	4	Ordinary
	Australia Ltd Australia Ltd	Buy	609'01	1,096	Ordinary
	Australia Ltd	Buy	9,448	976	Ordinary
	Suctralia 1 to	Buy	155	16	Ordinary
	יייים וויים ביים	Buy	852	88	Ordinary
	Australia Ltd	Buy	7,705	967	Ordinary
	Australia Ltd	Buy	6,737	969	Ordinary
	Australia Ltd	Buy	7,947	128	Ordinary
	Australia Ltd	Buy	165	17	Ordinary
	Australia Ltd	Buy	2,556	797	Ordinary
	Australia Ltd	Buy	14,307	1,478	Ordinary
	Australia Ltd	Buy	1,773	803	Ordinary
	Australia Ltd	Buy	155	16	Ordinary
	Australia Ltd	Buy	10	1	Ordinary
	Australia t.td	Buy	7,202	744	Ordinary
	Australia Ltd	Buy	16,998	1,756	Ordinary
	Australia Ltd	Buy	3,872	400	Ordinary
	Australia Ltd	Buy	94,661	6/1/6	Ordinary
	Australia Ltd	Buy	3,330	344	Ordinary
	Securities Australia Ltd	Buy	435,600	45,000	Ordinary
	Australia Ltd	Buy	16,166	1,670	Ordinary
	Australia Ltd	Buy	445	46	Ordinary
	Australia Ltd	Buy	13,087	1,352	Ordinary
į	Australia Ltd	Buy	2,052	212	Ordinary
	Australia Ltd	Buy	7,212	745	Ordinary
Г	Australia Ltd	Buy	620	64	Ordinary
28-Jul-14 UBS Securities Australia Ltd	Australia Ltd	Buy	155	16	Ordinary
Г	Securities Australia Ltd	Buv	6.253	646	Ordinary
Ī	Australia I td	Bary	CPR	19	Ordinary
08-Jul-14 UBS Securities Australia Ltd	Australia Ltd	Buy	3 477	354	Ordinary
Ť	Australia (†d	Bin	5003	300	Ordinany
T	Action of the	9.0	2,023	503	Colling
Ť	Australia Ltd	bug.	2,139	177	Orginary
T	Australia Liu	ono.	3,0,6	218	Crainary
T	Australia Ltd	aris	1,452	150	Ordinary
UB-Jul-14 UBS Securities Australia Ltd	Australia Ltd	Buy	242	25	Ordinary
1	Australia Ltd	Buy	284,127	29,352	Ordinary
	Australia Ltd	Buy	397	41	Ordinary
08-Jul-14 UBS Securities Australia Ltd	Australia Ltd	Buy	1,220	126	Ordinary
	Australia Ltd	Buy	869'8	382	Ordinary
	Securities Australia Ltd	Buy	01	-	Ordinary
Jul-14 UBS Securities Australia Ltd	Australia Ltd	Buy	1,084	112	Ordinary
Ī	Australia Ltd	Buy	3.107	321	Ordinary
08-Jul-14 UBS Securities Australia Ltd	Australia Ltd	Buv	1491	154	Ordinary
Ī	Australia Ltd	Briv	2 952	305	Ordinary
Ī	Australia I td	Buy	155	191	Ordinary
T	ustralia I to	in a	073 N	527	Ordinary
08-1ul-14 I IRS Securities Australia I to	uistralia I td	A Maria	157.2	503	Ordinary
T	the state of the s	, , ,	10/20	766	Oldinary Services
T	Australia Liu	ing	02C,1	/5/	Ordinary
Ť	Vostigila Lio	ing o	8,025	678	Ordinary
Ť	Vostralia Ltd	any.	12,720	1,314	Ordinary
OS-Jul- (4 UBS Securities Australia Ltd	vustralia Ltd	Buy	1,955	202	Ordinary
Ť	vustralia Ltd	Buy	899	69	Ordinary
Ť	vustralia Ltd	Buy	4,501	465	Ordinary
	kustralia Ltd	Buy	4,443	459	Ordinary
	vustralia Ltd	Buy	165	17	Ordinary
Ī	vustralia Ltd	Buy	1,588	164	Ordinary
08-Jul-14 UBS Securities A	sustralia Ltd	Buy	8,741	606	Ordinary
	sustralia Ltd	Buy	0/0/6	937	Ordinary
Ī	vustralia Ltd	Buv	165	17	Ordinary
On 11/14 I IDO Contribute Australia 14d	C+ Cicator	, in a	2112	7	Civinian

80 80 80 80 80 80 80 80 80 80 80 80 80 8	
	5 A A A A A A A A A A A A A A A A A A A
4444444444444444	70 A A A A A A A A A A A A A A A A A A A
	1
33333333333333333333333333333	
A A A A A A A A A A A A A A A A A A A	
A A A A A A A A A A A A A A A A A A A	
<u> </u>	
3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	
9997444747441	
44444444444	10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
11111111111	\(\text{Aris} \) \(Ari
111111111	9 And And And And And And And And And And
11111111	6 90 90 90 90 90 90 90 90 90 90 90 90 90
	80,0 80,0 80,0 80,0 80,0 80,0 80,0 80,0
111111	80 y y 80 y y y y y y y y y y y y y y y
111111	80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9
1 1 1	8 U.y 8 B.u.y 8 B.u.y 8 B.u.y 8 B.u.y 8 B.u.y
1 1 1 1 1	8 by 8 by 8 by 8 by 8 by 8 by 8 by
1111	809 809 809 809 809 809 809
l. l. l.	Buy Buy Buy Buy Buy Buy
1.1.	8uy 8uy 8uy 8uy
l.	Buy Buy Buy
	Buy
١,	Bu
2	
3	8
ģ	
È	
3	ω.
à.	18
ļ.	18
3	18
ج	38
۱,	ηθ
≥	96
L	58
L	-B
١.	N.B.
1.	600
J	08
↲	Bu
_	Bu
١,	PΩ
١	18
١,	à
4	
ş	1
3	8
2	ä
ا	
ş	
ξģ	
BÇ	
	\(\text{Ang} \) \(\te

08-Jul-14	1 DC Countition Australia 1+d	, and			
71-Jul-80	I IRS Societies Australia Ltd	200	666,1	381	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	426	44	Ordinary
NS 11/1 1/1	I IDO Contration Australia I to	, no	203		or all a
1 1 1 2	I IDC Cocurition Australia Ltd	600	360 1	2/2	Ordinary
1	OBS Securities Australia Liu	Buy	066,1	2007	Ordinary
#1-inc-00	ODS Securities Abstraria Litic	boy	50/	6/	Crainary
-Jul-14	UBS Securities Australia Ltd	Buy	1,984	502	Ordinary
08-JUI-14	UBS Securities Australia Ltd	Buy	7,638	/88	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	2,904	300	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	8,731	905	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	35,303	3,647	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buy	1,529	158	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
28-Jul-14	UBS Securities Australia Ltd	Buy	11,984	1,238	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	571	59	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buv	1.162	120	Ordinary
-Jul-14	UBS Securities Australia I td	Buy	3 407	357	Ordinary
08-1-1-14	I IBS Securities Australia I to	\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	116	12	Ordinan
28-1-1-14	I IBC Coourities Australia 14d	200	2 547	090	Ordinary
D8-1-14	LIBS Securities Australia 14d	200	358	800	Ordingo
DR-1-1-14	I BC Counties Australia 11d	e e	679	2 2	Ording
11 17	tipo contribo Australia Lia	909	700	2 5	Cicinary
	Lan Committee Australia Ltd	900	100	į ç	Oldinally
11 1 40	1 DC Counties Australia Ltd	600	12/,4	op oc	Ordinal y
- 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	Cos securities Australia Liu	960	172	07	Cidinary
20-301-14	CBS Securities Australia Lid	eu.	2,504	000	Ordinary
+1-ln-00	Obs Securities Australia Ltd	ond.	911	71	Crainary
10-10-	Upo Securities Australia Ltd	Buy	000	79	Orainary
08-101-14	UBS Securities Australia Ltd	Buy	862	60	Ordinary
J8-JUI-14	UBS securities Australia Ltd	Buy	2,856	295	Ordinary
8-Jul-14	UBS Securities Australia Ltd	Buy	2,827	292	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buy	194	50	Ordinary
8-Jul-14	UBS Securities Australia Ltd	Buy	1,220	126	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buy	339	35	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buv	2.284	236	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buv	1.752	181	Ordinary
-Juj-14	IUBS Securities Australia Ltd	Buv	2,759	285	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	2 739	283	Ordinary
11.1.1	I IDC Countries Australia 144	Po d	352	2	Ordinary Ordinary
11 - Inc-00	DDS Securities Australia Liu	pay	507	600	Clemary
-701-14	UBS Securities Australia Ltd	Buy	169.7	8/7	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	2,710	280	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	7,928	819	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	2,449	253	Ordinary
28-Jul-14	UBS Securities Australia Ltd	Buy	2 604	269	Ordinary
-1.11-14	I IBS Securities Australia I to	2 8	513	E2	Ordinary
1 1 1 00	Language Authority and	603	1001	3	O CITE O
1 1	Obs Securities Australia Ltd	gnà	4,801	0.5	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	5,643	283	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	4,279	442	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	2,585	267	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buy	2,459	254	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buy	2,236	231	Ordinary
08-Jul-14	UBS Securities Australia itd	Buy	17 705	1 879	Ordinary
11-17	1 IRC Constitute Australia 14d	S. S.	6003	623	Ordingo
08-1-1-14	1 BC Counties Australia 1+4	Britis	2 381	270	Cipiling C
	Con peculities Australia Etc	ouy e	1977	757	Oldinal y
-101-14	UBS Securities Australia Ltd	Buy	1,316	36	Ordinary
28-Jul-14	UBS Securities Australia Ltd	Buy	2,430	251	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	3,756	388	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buy	3,010	311	Ordinary
08-Jul-14	S S	Buy	920	95	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buv	3.127	323	Ordinary
08-101-14	UBS Securities Australia Ltd	Buy	2.991	308	Ordinary
08-11-17	LIRC Countries Australia Ltd	P	20115	970.6	Ordinary
Inf-	UBS Securities Australia Ltd	Ruy	20,115	2,0/0	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	4,927	209	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	3,872	400	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buy	2,420	250	Ordinary
11.1.1				2	5 5
	The contract of the contract o	i d	100.0	245	Cierro C
US-JUI-14	UBS Securities Australia Ltd	Buy	2,081	215	Ordinary

By 7,701 279 By 7,173 279 By 1,162 2.0 By 2,233 2.40 By 3,82 2.40 By 3,82 2.40 By 3,82 2.03 By 2,62 2.0 By 2,62 4.0 By 2,62	UBS Securities Australia Ltd UBS Securities Australia Ltd	80.0 80.0	2,701 4,666 1,113 5,876 2,817 1,162 1,162 1,162 2,033 2,033 2,033 2,033 1,04	279 482 482 607 29 190 110 204 400 400 204 400 1140 1140 1141 11302 1302 1303 1303 1408 1144 1144 1144 1144 1144 1144 1144	Ordinary Ordinary
Buy 4,4173 741 Buy 4,6178 421 Buy 1,113 115 Buy 1,112 1,20 Buy 1,112 1,20 Buy 1,112 1,20 Buy 1,112 1,20 Buy 2,233 2,10 Buy 3,847 393 Buy 4,821 4,99 Buy 2,532 2,03 Buy 3,847 3,99 Buy 1,757 2,66 Buy 1,764 1,45 Buy 1,757 2,57 Buy 1,764 1,45 Buy 1,764 1,45 Buy 1,344 1,44 Buy 1,757 2,57 Buy 1,764 1,45 Buy 1,344 1,44 Buy 1,344 1,44 Buy 1,344 1,44 Buy 1,244 800 <tr< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>8 Buy 8 Buy</td><td>7,173 7,173 7,173 5,876 1,162 2,323 2,323 3,872 2,323 2,573 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,757 2,</td><td>482 607 29 29 408 408 408 408 408 204 204 204 204 204 204 204 204</td><td>Ordinary Ordinary /td></tr<>	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 Buy 8 Buy	7,173 7,173 7,173 5,876 1,162 2,323 2,323 3,872 2,323 2,573 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,575 2,757 2,	482 607 29 29 408 408 408 408 408 204 204 204 204 204 204 204 204	Ordinary Ordinary
Buy 4,666 482 Buy 1,113 115 Buy 2,816 607 Buy 2,816 607 Buy 2,823 240 Buy 2,323 240 Buy 2,527 2,88 Buy 1,984 1,164 Buy 1,295 1,260 Buy 1,275 1,285 Buy 1,240 1,291 Buy 1,240 1,251 Buy 1,240 1,251 Buy 2,344 1,354 Buy 2,344 1,354	UBS Securities Australie Ltd UBS Securities Australie Ltd	8 80 9 80 9 80 9 80 9 9 9 9 9 9 9 9 9 9	1,113 5,876 1,162 2,323 4,821 3,804 3,804 3,804 1,757 1,757 1,757 1,757 1,757 1,757 1,757 1,757 2,	482 1150 120 120 120 120 120 120 120 12	Ordinary Ordinary
Buy 5,173 115 Buy 5,876 60 Buy 1,162 1,20 Buy 1,162 1,20 Buy 1,162 1,20 Buy 2,332 2,40 Buy 2,333 2,10 Buy 2,532 2,66 Buy 1,297 2,03 Buy 1,297 2,04 Buy 1,297 2,04 Buy 1,297 2,04 Buy 1,297 2,04 Buy 1,297 2,03 Buy 1,297 2,04 Buy 1,297 2,04 Buy 1,297 2,03 Buy 1,297 2,04 Buy 1,347 1,14 Buy 1,347 1,14 Buy 1,347 1,14 Buy 1,297 2,03 Buy 1,297 2,03 Buy 1,297 2,04 <tr< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>8 buy 8 buy</td><td>1,13 1,162 2,876 1,162 2,323 3,872 3,872 3,872 3,872 3,872 1,570 1,5</td><td>115 117 117 117 117 117 117 117</td><td>Ordinary Ordinary /td></tr<>	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 buy 8 buy	1,13 1,162 2,876 1,162 2,323 3,872 3,872 3,872 3,872 3,872 1,570 1,5	115 117 117 117 117 117 117 117	Ordinary Ordinary
Buy 5.876 60 Buy 5.876 60 Buy 1.162 1.20 Buy 1.182 1.20 Buy 1.182 1.20 Buy 1.182 1.20 Buy 1.823 2.10 Buy 2.333 2.10 Buy 1.975 2.04 Buy 1.975 2.04 Buy 1.800 1.800 Buy 1.800 1.800 Buy 1.304 1.144 Buy 1.304 1.145 Buy 1.300 1.800 Buy 1.304 1.145 Buy 1.304 1.145 Buy 1.304 1.444 Buy 1.304 1.444 Buy 1.304 1.444 Buy 1.304 1.455 Buy 1.304 1.450 Buy 1.304 1.444 Buy 1.304 1.444	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	28176 28176 2817 2817 2817 2818 2818 2818 2757 2757 2757 2757 2757 2757 2757 275	29 240 400 400 400 240 400 204 204	Ordinary Ordinary
Buy 2,81 2,9 Buy 1,162 1,0 Buy 2,322 240 Buy 2,322 240 Buy 2,322 240 Buy 2,322 240 Buy 2,632 490 Buy 2,572 262 Buy 2,573 2.0 Buy 2,757 2.6 Buy 1,104 1,14 Buy 1,3475 1,382 Buy 2,757 285 Buy 1,3475 1,43 Buy 1,3475 1,43 Buy 1,5475 1,382 Buy 1,5475 1,382 Buy 1,5475 1,382 Buy 1,5475 1,48 Buy 1,5475 2,58 Buy 1,5475 2,68 Buy 1,5475 3,68 Buy 1,242 4,68 Buy 1,5475 1,425	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	1,162 2,323 4,821 3,804 2,033 2,503 2,575 2,575 2,575 2,757 2,757 2,757 2,757 2,757 2,757 2,757 2,757 2,757 2,757 1,394 1,398 1,340 1,540	120 120 120 408 408 408 208 208 208 208 208 208 208 208 208 2	Ordinary Ordinary
Buy 1,152 240 Buy 2,323 240 Buy 2,323 240 Buy 3,872 489 Buy 2,673 2,10 Buy 2,575 264 Buy 2,575 266 Buy 1,104 1,14 Buy 1,275 265 Buy 1,275 268 Buy 1,275 285 Buy 1,360 1,860 Buy 1,604 1,145 Buy 1,604 1,146	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	1,162 1,162 1,162 1,203 1,570 1,	240 498 408 409 393 392 204 206 405 11860	Ordinary Ordinary
Buy 4,223 4,24 Buy 4,223 4,24 Buy 3,872 4,09 Buy 3,872 400 Buy 2,033 210 Buy 2,033 210 Buy 2,575 266 Buy 2,757 285 Buy 2,757 285 Buy 2,757 285 Buy 2,757 285 Buy 1,394 1,145 Buy 1,347 1,392 Buy 1,347 1,392 Buy 1,347 1,392 Buy 1,347 1,392 Buy 1,347 1,45 Buy 1,347 1,45 Buy 1,340 4,68 Buy 1,340 4,68 Buy 1,340 4,60 Buy 1,342 3,74 Buy 1,342 3,74 Buy 1,342 3,74 <tr< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9</td><td>2.33 4.821 3.872 3.804 3.804 2.033 2.573 1.575 2.757 2.757 2.757 2.757 2.757 2.757 2.757 2.757 2.757 3.886 3.998 8.886 3.998 8.886 3.998 13.998</td><td>400 400 400 204 204 204 205 206 206 206 206 206 206 206 206 206 206</td><td>Ordinary Ordinary /td></tr<>	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	2.33 4.821 3.872 3.804 3.804 2.033 2.573 1.575 2.757 2.757 2.757 2.757 2.757 2.757 2.757 2.757 2.757 3.886 3.998 8.886 3.998 8.886 3.998 13.998	400 400 400 204 204 204 205 206 206 206 206 206 206 206 206 206 206	Ordinary Ordinary
Buy 3,621 409 Buy 3,804 400 Buy 3,804 393 Buy 2,033 2,03 Buy 2,575 204 Buy 7,570 7,87 Buy 7,570 7,85 Buy 1,104 1,14 Buy 2,757 2,85 Buy 3,875 4,85 Buy 2,757 2,85 Buy 2,757 2,85 Buy 3,998 4,18 Buy 3,643 4,68 Buy 2,757 2,85 Buy 1,347 1,45 Buy 1,347 1,45 Buy 2,63 4,68 Buy 1,347 1,45 Buy 1,347 1,45 Buy 1,347 1,45 Buy 1,342 3,24 Buy 1,342 3,24 Buy 1,342 3,24	UBS Securities Australia Ltd UBS Securities Australia Ltd	80.0 80.0	3,872 3,874 2,575 1,975 1,1975 2,575 2,575 2,757	490 400 782 782 782 785 785 785 785 785 785 785 785 785 785	Ordinary Ordinary
Buy 3,677 400 Buy 3,677 400 Buy 2,757 2,03 Buy 2,757 2,04 Buy 1,975 2,04 Buy 1,675 2,04 Buy 1,704 1,14 Buy 2,757 2,26 Buy 2,757 2,26 Buy 2,757 2,85 Buy 2,787 2,87 Buy 2,63 3,00 Buy 2,743 80 Buy 2,743 80 <t< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9</td><td>3,804 2,033 1,570 1,975 1,975 1,104 1,104 1,394 1,394 1,394 1,340 1,540</td><td>393 393 200 200 200 200 200 1,860 1,860 1,860 1,860 1,860 1,186 1,</td><td>Ordinary Ordinary /td></t<>	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	3,804 2,033 1,570 1,975 1,975 1,104 1,104 1,394 1,394 1,394 1,340 1,540	393 393 200 200 200 200 200 1,860 1,860 1,860 1,860 1,860 1,186 1,	Ordinary Ordinary
Buy 2,0204 329 Buy 2,0304 329 Buy 2,033 210 Buy 2,757 264 Buy 2,757 285 Buy 3,826 918 Buy 1,347 1,450 Buy 1,247 1,450 Buy 2,510 4,501 Buy 2,520 4,501 Buy 2,520 4,501	UBS Securities Australia Ltd UBS Securities Australia Ltd	804 804 804 804 804 804 804 804 804 804	2,034 2,033 2,575 2,575 2,575 2,575 2,757 2,757 2,757 2,757 2,757 3,875 1,398	200 204 204 45 45 45 40 40 40 40 40 40 40 40 40 40 40 40 40	Ordinary Ordinary
Buy 7,203 210 Buy 7,575 204 Buy 1,104 1,14 Buy 2,575 266 Buy 2,757 285 Buy 1,342 810 Buy 1,342 810 Buy 1,540 1,541 Buy 4,530 4,501 Buy 1,540 1,541 Buy 1,540 1,541 Buy 2,528 3,425 3,542 Buy 2,544 800 Buy 2,544 800 </td <td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td> <td>8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9</td> <td>2,033 2,575 1,975 436 1,104 1,805 2,757 2,75</td> <td>782 782 782 785 785 786 786 786 786 786 786 787 887 77 77 77 788 788</td> <td>Ordinary Ordinary /td>	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	2,033 2,575 1,975 436 1,104 1,805 2,757 2,75	782 782 782 785 785 786 786 786 786 786 786 787 887 77 77 77 788 788	Ordinary Ordinary
By 1,570 782 By 1,570 782 By 1,570 266 By 1,570 266 By 1,570 2,575 2.66 By 1,600 1,44 45 By 1,600 1,800 1,860 By 1,757 2.85 By 1,394 1,14 By 1,394 1,14 By 1,394 1,14 By 1,394 1,14 By 1,3475 1,392 By 1,3475 1,392 By 1,3475 1,392 By 1,3475 1,392 By 1,3475 1,352 By 1,3475 1,352 By 1,3475 1,352 By 1,3475 1,450 By 1,3475 1,450 By 1,3475 1,441 By 1,243 1,441 By	UBS Securities Australia Ltd UBS Securities Lustralia Ltd	89.9 80.0	7,570 1,975 1,104 1,104 1,104 1,104 1,104 1,108	204 206 1141 11860 285 285 285 285 285 286 144 918 11392 11392 11392 11591 1145 1145 1145 1145 1145 1145 114	Ordinary Ordinary
Byy 1,975 264 Byy 2,757 265 By 1,104 1,14 By 1,104 1,14 By 1,704 1,14 By 2,757 285 By 1,342 30 By 1,342 31 By 1,342 4,58 By 1,344 1,43 By 1,344 1,43 By <t< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>89.0 89.0</td><td>1,975 436 1,104 1,104 1,104 1,394 1,</td><td>204 206 206 206 206 206 206 206 206 206 206</td><td>Ordinary Ordinary /td></t<>	UBS Securities Australia Ltd UBS Securities Australia Ltd	89.0 89.0	1,975 436 1,104 1,104 1,104 1,394 1,	204 206 206 206 206 206 206 206 206 206 206	Ordinary Ordinary
Buy 2,575 266 Buy 436 436 436 Buy 1,104 1,144 <t< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>89.9 80.9</td><td>2575 436 1,104 18,005 2,757 2,757 2,757 2,757 13,405 13,405 15,401 15,401 15,401 16,40</td><td>266 1,860 1,860 1,860 1,860 1,860 1,180 1,591 1,</td><td>Ordinary Ordinary /td></t<>	UBS Securities Australia Ltd UBS Securities Australia Ltd	89.9 80.9	2575 436 1,104 18,005 2,757 2,757 2,757 2,757 13,405 13,405 15,401 15,401 15,401 16,40	266 1,860 1,860 1,860 1,860 1,860 1,180 1,591 1,	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
By 436 45 By 1,104 1,114 By 1,104 1,114 By 1,104 1,114 By 2,757 285 By 3,826 9,18 By 3,826 9,18 By 3,644 810 By 4,541 1,45 By 4,542 324 By 4,544 800 By 4,544 800 By 4,542 324 By 4,542 3244 By 4,54	UBS Securities Australia Ltd UBS Securities Australia Ltd	80y 80y 80y 80y 80y 80y 80y 80y	1,104 18,005 18,005 2,757 2,757 2,757 1,394 1,396 1,39	45 11860 1860 400 400 1144 1132 11332 11332 11332 11343 1145 1145 1145 1145 1145 1145 114	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 1 (104 114 Buy 1, 104 1, 104 Buy 2,757 285 Buy 2,757 285 Buy 2,757 285 Buy 2,757 285 Buy 1,342 413 Buy 1,342 413 Buy 15,401 1,501 Buy 15,401 1,501 Buy 15,401 1,501 Buy 45,201 45,01 Buy 45,201 45,01 Buy 45,201 45,01 Buy 45,201 45,01 Buy 45,204 45,01 Buy 45,204 45,01 Buy 45,204 45,01 Buy 45,204 1,44 800 Buy 1,342 80 Buy 2,342 3,425 3,425 Buy 2,342 3,425 3,425 Buy 2,903 3,00	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 by 8 by	1,104 1,104 1,104 1,1084 1,1084 1,1084 1,1084 1,1084 1,1084 1,1084 1,1084 1,1084 1,1084 1,1084 1,1084	114 285 285 285 400 144 918 918 1392 810 1591 1591 1591 1592 817 87 87 87 87 87	Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary
Buy 18,005 1,860 Buy 2,757 285 Buy 8,886 918 Buy 3,998 413 Buy 1,3475 1,392 Buy 1,3475 1,392 Buy 2,648 918 Buy 7,841 87 Buy 7,842 87 Buy 7,843 4,688 Buy 1,540 4,591 Buy 45,398 4,688 Buy 45,398 4,689 Buy 2,048 4,591 Buy 2,048 4,591 Buy 2,048 2,005 Buy 1,404 1,457 Buy 1,457 1,467 Buy 1,404 1,457 Buy 1,2,398 2,005 <	UBS Securities Australia Ltd UBS Securities Australia Ltd	86,4 80,4	18,005 2,757 2,757 2,757 1,394 8,886 13,495 13,495 13,401 15,401 15,401 15,401 10,84	1,860 285 285 286 400 144 918 918 132 132 1,592 810 1,592 810 1,593 1,145 1,145 1,145 81 81 1,145 1,145 1,145 81 81 1,145 1,14	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 2,757 286 Buy 2,757 286 Buy 3,757 285 Buy 3,757 285 Buy 3,757 285 Buy 3,757 285 Buy 3,757 400 Buy 3,757 400 Buy 1,347 1,327 Buy 1,344 810 Buy 1,5401 1,591 Buy 1,6401 1,591 Buy 2,530 4,581 Buy 2,530 4,581 Buy 2,542 82 Buy 2,542 82 Buy 2,642 87 Buy 2,642 87 Buy 2,643 87 Buy 2,643 87 Buy 2,644 800 Buy 2,642 87 Buy 2,643 80 Buy 2,943 3,642 B	UBS Securities Australia Ltd UBS Securities Australia Ltd	80.7 80.7	2,757 2,757 3,872 1,394 1,396 1,396 1,396 1,475 1,841 1,5401 1,54	285 400 400 400 1144 11392 810 810 810 1,591 1,591 1,145 1,1	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 2,727 285 Buy 2,777 2,677 2,677 2,677 2,677 2,677 2,677 2,677 2,677 2,687 400 80 80 81 82 81 81 82 81 82 82 81 81 82	UBS Securities Australia Ltd UBS Securities Australia Ltd	80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9	2,757 3,872 1,394 8,886 3,998 13,998 7,841 68 13,10 11,004 11,004	285 400 144 918 1392 810 7 7 7 7 1,591 1,591 1,591 1,591 1,45 1,468 4,688 87 87 87 87 87 88 7	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
By 5,73 400 By 3,872 400 By 1,394 144 By 1,394 144 By 1,394 1,475 By 1,394 1,332 By 1,3475 1,332 By 1,475 1,332 By 6,88 918 By 6,88 1,145 By 1,1084 1,145 By 4,530 4,501 By 1,434 1,444	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	3.872 1,394 8,886 3,998 13,475 7,841 68 310 115,401 11,084	400 400 1144 918 11,392 810 72 7 7 7 7 7 7 1,591 1,591 1,458 4,688 4,688 8 8 8 8 1,448 1,4	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 1,394 140 Buy 1,394 144 Buy 1,394 144 Buy 1,394 141 Buy 1,347 1810 Buy 1,447 1810 Buy 15,475 1,392 Buy 15,401 1,591 Buy 15,401 1,591 Buy 15,401 1,591 Buy 15,401 1,591 Buy 1,623 4,501 Buy 4,538 4,501 Buy 4,570 4,501 Buy 4,570 4,501 Buy 4,570 4,501 Buy 4,570 4,501 Buy 2,570 4,501 Buy 1,544 800 Buy 1,365 1,444 Buy 1,366 1,444 Buy 1,366 1,444 Buy 1,366 1,444 Buy 1,366 1,444 <td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td> <td>8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by</td> <td>3,872 1,394 8,886 3,998 13,475 7,841 68 310 15,401 15,401 11,084</td> <td>1400 1410 11302 11302 11301 11501 11501 1145 1145 1145 1145 1145</td> <td>Ordinary Ordinary /td>	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by	3,872 1,394 8,886 3,998 13,475 7,841 68 310 15,401 15,401 11,084	1400 1410 11302 11302 11301 11501 11501 1145 1145 1145 1145 1145	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
By 1894 184 By 8,846 144 By 8,846 113 By 1,3475 1,392 By 1,3475 1,392 By 1,3475 1,392 By 1,3475 1,392 By 1,641 1,322 By 1,540 1,541 By 1,540 1,541 By 1,542 8.7 By 4,532 4,638 By 4,532 4,638 By 4,532 4,532 By 4,532 4,532 By 4,532 4,532 By 1,433 3,03 By 1,434 1,454 By 1,434 1,457 By 1,434 1,457 By 1,444 1,457 By 1,457 1,447 By 1,444 1,457 By 1,444 1,457	UBS Securities Australia Ltd UBS Securities Australia Ltd	80.y 80.y 80.y 80.y 80.y 80.y 80.y 80.y	1,394 8,886 3,998 13,475 7,841 15,401 15,401 11,084	144 918 413 1,392 810 7 7 1,591 1,591 1,445 4,688 4,688	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 8.886 918 Buy 3.986 918 Buy 7,841 810 Buy 26 7 Buy 26 7 Buy 2,438 4,688 Buy 8,238 9,5 Buy 8,238 9,7 Buy 2,603 4,501 Buy 2,1,756 80 Buy 2,428 9,7 Buy 2,428 9,0 Buy 16,341 1,689 Buy 2,042 80 Buy 2,043 9,0 Buy 1,342 80 Buy 1,342 1,444 Buy 1,342 1,444 Buy 1,342 1,447 Buy 1,366 1,427 Buy	UBS Securities Australia Ltd UBS Securities Australia Ltd	80.y 80.y 80.y 80.y 80.y 80.y 80.y 80.y	8,886 3,998 13,475 7,441 68 68 11,5,401 11,084 7,5,300	918 1,392 81,092 1,392 27 27 27 27 1,145 1,145 1,145 6,23	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Bby 3,998 413 Buy 1,3475 1,322 Buy 7,841 810 Buy 7,841 810 Buy 7,841 810 Buy 11,641 27 Buy 11,642 87 Buy 11,084 1,531 Buy 45,380 4,688 Buy 45,380 4,688 Buy 45,380 4,688 Buy 45,380 4,689 Buy 4,5370 4,501 Buy 210,489 2,756 Buy 21,248 800 Buy 7,744 800 Buy 1,385 3,499 Buy 1,385 3,499 Buy 1,386 1,41 Buy 1,386 1,44 Buy 1,386 1,402 Buy 1,386 1,41 Buy 1,386 1,41 Buy 1,366 1,42	UBS Securities Australia Ltd UBS Securities Australia Ltd	80y 80y 80y 80y 80y 80y 80y 80y 80y 80y	3,998 13,475 7,841 68 310 15,401 15,401 11,004	413 1,392 810 7 7 1,591 1,145 4,688 87 623	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
By 13,475 1,332 Buy 6,441 810 Buy 6,84 7,841 810 Buy 6,84 7,145 81 Buy 1,1084 1,145 82 Buy 1,1084 1,145 82 Buy 26,13 6,23 6,23 Buy 4,530 4,501 82 Buy 4,530 4,501 82 Buy 4,520 4,501 82 Buy 4,520 4,501 82 Buy 4,520 4,501 82 Buy 1,634 1,452 80 Buy 2,744 800 80 Buy 3,853 3,00 80 Buy 1,244 1,441 1,441 Buy 1,365 1,442 1,441 Buy 1,365 1,442 1,442 Buy 1,360 1,442 1,442 Buy 1,360	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by	13,475 7,841 68 83 310 15,401 261 11,004	1,392 810 7 7 32 1,591 27 27 27 27 4,688 87 87	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 7,841 810 Buy 568 7 Buy 510 32 Buy 15,401 1,591 Buy 15,401 1,591 Buy 15,401 1,591 Buy 45,380 4,688 Buy 45,380 4,688 Buy 45,370 4,501 Buy 4,570 4,501 Buy 2,570 4,501 Buy 16,341 1,689 Buy 16,341 1,689 Buy 1,344 800 Buy 1,345 1,760 Buy 1,344 800 Buy 1,345 1,444 Buy 1,366 1,475 Buy 2,914 1,475	UBS Securities Australia Ltd UBS Securities Australia Ltd	8 by 8 by 8 by 8 by 8 by 8 by 8 by 8 by	7,841 68 310 15,401 261 11,084	810 7 32 1,591 27 27 27 1,145 4,688 87 87	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 66 7 Buy 15,401 1,591 Buy 15,401 1,591 Buy 45,380 4,688 Buy 45,380 4,601 Buy 45,370 4,501 Buy 45,70 4,501 Buy 45,70 4,501 Buy 16,341 1,689 Buy 1,634 1,689 Buy 1,634 1,644 Buy 1,345 1,410 Buy 1,344 1,457 Buy 1,356 1,41 Buy 1,360 1,41 Buy 1,360 1,457 Buy 1,376 1,457 Buy 1,4104 1,457 Buy 1,4104 <	UBS Securities Australia Ltd UBS Securities Australia Ltd	84.9 84.9 84.9 84.9 84.9 84.9 84.9 84.9	68 310 15,401 261 11,084	7 32 32 1,591 27 1,145 4,688 87 87	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 310 32 Buy 15,401 1,591 Buy 261 27 Buy 261 27 Buy 45,380 4,688 Buy 84,380 4,688 Buy 84,280 4,501 Buy 8,570 4,501 Buy 21,756 33,84 Buy 3,843 30,63 Buy 16,341 1,689 Buy 3,744 800 Buy 3,742 30 Buy 3,745 30 Buy 7,745 80 Buy 1,689 30 Buy 1,742 80 Buy 1,742 80 Buy 1,360 14 Buy <td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td> <td>Buy Buy Buy Buy Buy Buy Buy Buy</td> <td>310 15,401 261 11,084</td> <td>32 1,591 27 27 1,145 4,688 87 623</td> <td>Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary</td>	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy Buy Buy Buy Buy Buy Buy	310 15,401 261 11,084	32 1,591 27 27 1,145 4,688 87 623	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 15,401 1,591 Buy 261 27 Buy 45,380 4,688 Buy 45,380 4,688 Buy 6,031 623 Buy 6,031 623 Buy 4,501 6,031 Buy 21,048 2,756 Buy 1,6241 1,689 Buy 2,903 300 Buy 7,744 800 Buy 7,744 800 Buy 7,744 800 Buy 7,765 1,760 Buy 1,365 1,424 Buy 1,366 1,424 Buy 1,366 1,425 Buy 1,366 1,425 Buy 1,366 1,424 Buy 1,366 1,422 Buy 1,366 1,422 Buy 1,366 1,422 Buy 1,366 1,422 Buy 2,314 1,366	UBS Securities Australie Ltd UBS Securities Australia Ltd	Buy Buy Buy Buy Ruy Ruy	15,401	1,591 27 27 1,145 4,688 87 87 623	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 261 27 Buy 45,380 4,688 Buy 45,380 4,681 Buy 6,031 6,23 Buy 43,570 4,501 Buy 43,570 4,501 Buy 43,570 4,501 Buy 16,341 1,585 Buy 16,341 1,689 Buy 3,823 3,90 Buy 3,823 3,499 Buy 3,823 3,499 Buy 3,828 3,499 Buy 3,864 1,41 Buy 1,743 1,544 Buy 1,386 1,41 Buy 1,398 1,457 Buy 1,398 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 1,4104 1,457 Buy 1,4104 1	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy Buy Buy Rivy	11,084	27 1,145 4,688 87 87 623	Ordinary Ordinary Ordinary Ordinary Ordinary
Buy 11,084 1,145 Buy 45,380 4,688 Buy 84,2 87 Buy 8,570 4,501 Buy 6,031 6,23 Buy 6,031 6,23 Buy 9,438 9,75 Buy 7,748 800 Buy 7,744 800 Buy 7,744 800 Buy 7,742 80 Buy 7,744 800 Buy 7,744 800 Buy 7,742 80 Buy 1,760 80 Buy 1,363 141 Buy 1,365 142 Buy 1,365 145 Buy 2,914 301 Buy 2,914 301 Buy 2,914 457 Buy 2,914 457 Buy 2,914 445 Buy 2,914 445 B	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy Buy	11,084	1,145 4,688 87 87 623	Ordinary Ordinary Ordinary Ordinary
Buy 45,380 4,688 Buy 842 87 Buy 6,031 6,23 Buy 6,031 6,53 Buy 210,489 2,570 4,501 Buy 210,489 2,573 304 Buy 2,043 300 300 Buy 7,444 800 Buy 7,444 800 Buy 7,604 300 Buy 7,644 800 Buy 1,385 3,499 Buy 2,903 3,000 Buy 1,385 1,41 Buy 1,385 1,44 Buy 1,386 2,005 Buy 1,386 1,42 Buy 1,386 1,42 Buy 1,376 1,42 Buy 1,376 1,42 Buy 2,31 2,31 Buy 2,31 2,31 Buy 2,31 4,42 Buy<	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy	000 1.	4,688	Ordinary Ordinary Ordinary
Buy 642 87 Buy 6031 623 Buy 6031 623 Buy 4570 4501 Buy 210,489 21,756 Buy 16,341 1,689 Buy 3,425 354 Buy 3,823 3,99 Buy 7,744 800 Buy 3,823 3,99 Buy 1,632 3,00 Buy 17,034 1,544 Buy 17,034 1,49 Buy 1,565 141 Buy 7,55 78 Buy 1,396 1,457 Buy 1,376 7,26 Buy 1,376 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 2,914 301 Buy 2,391 2,47 Buy 2,394 1,457 Buy 2,394 1,38 <t< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>Buy</td><td>47 4211</td><td>623</td><td>Ordinary Ordinary Ordinary</td></t<>	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	47 4211	623	Ordinary Ordinary Ordinary
Buy 6,031 6,23 Buy 6,034 6,23 Buy 4,570 4,501 Buy 2,10,489 27,756 Buy 16,341 1,689 Buy 1,744 800 Buy 7,744 800 Buy 7,744 800 Buy 7,744 800 Buy 1,365 1,560 Buy 1,366 1,440 Buy 1,366 1,44 Buy 1,366 1,45 Buy 2,178 2,54 Buy 2,176 2,25 Buy 2,176 2,25 Buy 2,176 2,54 Buy 2,914 301 Buy 2,914 301	UBS Securities Australia Ltd UBS Securities Australia Ltd	2 2	CVG	623	Ordinary Ordinary
Buy 45,570 4,501 Buy 45,570 4,501 Buy 210,489 21,756 Buy 16,341 1,88 Buy 3,425 354 Buy 2,744 800 Buy 2,744 800 Buy 2,80,800 60,000 Buy 18,421 1,564 Buy 1,94,421 1,560 Buy 1,346 1,141 Buy 1,346 1,457 Buy 1,398 2,005 Buy 1,4104 1,457 Buy 1,4104 1,457 Buy 2,178 2,58 Buy 2,178 2,18 Buy 2,814 301 Buy 2,814 301 Buy 2,814 1,43 Buy 2,814 1,43 Buy 2,814 1,43 Buy 2,814 1,43 Buy 2,814 1,	UBS Securities Australia Ltd UBS Securities Australia Ltd	2	740	620	Ordinary
Buy 2,270 4,270 Buy 21,256 9,75 Buy 21,256 364 Buy 3,425 354 Buy 3,425 354 Buy 2,703 300 Buy 3,823 3,499 Buy 3,823 3,409 Buy 17,744 800 Buy 1,503 1,444 Buy 1,362 141 Buy 1,365 141 Buy 1,365 141 Buy 1,365 141 Buy 1,365 142 Buy 1,365 142 Buy 1,376 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 2,914 301 Buy 2,914 301 Buy 2,391 2,47 Buy 2,394 1,38 Buy 2,394 3,47 <tr< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>600</td><td>0,000</td><td></td><td>Ordinary</td></tr<>	UBS Securities Australia Ltd UBS Securities Australia Ltd	600	0,000		Ordinary
Buy 210,488 978 Buy 210,489 21,756 Buy 16,341 1,689 Buy 3,425 354 Buy 7,444 800 Buy 2,903 3,00 Buy 2,803 3,499 Buy 140,421 15,444 Buy 1,3853 3,499 Buy 1,360 60,00 Buy 1,362 1,41 Buy 1,362 1,42 Buy 1,366 1,42 Buy 1,376 1,42 Buy 2,178 2,21 Buy 2,31 2,31 Buy 2,34 1,38 Buy 1,366 1,38 Buy 2,391 2,47 Buy 2,391 2,47	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	45,570	100,4	
Buy 21,756 Buy 16,341 1,589 Buy 3,425 354 Buy 2,903 300 Buy 2,803 300 Buy 3,863 3,499 Buy 17,037 1,760 Buy 17,037 1,760 Buy 13,398 2,005 Buy 1,365 141 Buy 1,365 141 Buy 1,370 1,457 Buy 1,370 1,457 Buy 1,378 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 2,914 301 Buy 2,914 301 Buy 2,814 1,43 Buy 1,386 46 Buy 1,384 1,13 Buy 1,386 45 Buy 1,386 45 Bu	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	9,438	975	Ordinary
Buy 16,341 1,689 Buy 2,903 3.04.2 Buy 7,744 80.0 Buy 7,744 80.0 Buy 7,744 80.0 Buy 3,3.653 3,4.69 Buy 19,3.98 2,4.44 Buy 1,3.68 1,4.60 Buy 1,3.68 1,4.7 Buy 1,3.65 1,4.7 Buy 1,3.65 1,4.7 Buy 2,1.76 2.25 Buy 2,1.76 3.01 Buy 2,3.16 3.01 Buy 2,3.16 3.01 Buy 2,3.16 3.01 Buy 2,3.16 3.01 Buy 2,3.16 3.01 Buy 2,3.16 3.01 Buy 2,3.16 3.01 Buy 2,3.16 3.01 Buy 2,5.16 3.01 Buy 2,5.16 3.01 Buy 2,5.16 3.01 Buy 2,5.16 2.02 Buy 2,5.17 2.01 Buy 3,5.17 3.01 Buy 3,5.17 3.0	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	210,489	21,756	Ordinary
By 3,425 354 By 7,744 800 By 2,903 300 By 2,903 300 By 33,853 3,499 By 33,823 3,499 By 14,242 1,760 By 1,7037 1,760 By 1,358 2,005 By 1,356 1,487 By 14,104 1,487 By 14,104 1,487 By 1,376 1,422 By 1,376 1,422 By 2,118 1,22 By 1,376 1,427 By 2,314 301 By 2,314 301 By 2,314 301 By 445 46 By 1,384 143 By 1,384 143 By 1,384 143 By 4,45 4,45 By <td>UBS Securities Australia Ltd UBS Securities Australia Ltd</td> <td>Buy</td> <td>16,341</td> <td>1,689</td> <td>Ordinary</td>	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	16,341	1,689	Ordinary
Buy 7,744 800 Buy 2,903 300 Buy 33,853 3,499 Buy 33,853 3,499 Buy 17,032 60,000 Buy 17,338 2,005 Buy 7,55 78 Buy 7,55 78 Buy 1,4104 1,457 Buy 1,736 1,457 Buy 1,736 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 1,376 1,457 Buy 2,914 301 Buy 2,914 301 Buy 2,914 46 Buy 2,914 46 Buy 1,386 1138 Buy 1,396 138 Buy 4,250 439 Buy 2,594 2,594 Buy 2,594 2,665	UBS Securities Australie Ltd UBS Securities Australia Ltd	Buy	3,425	354	Ordinary
Buy 2,903 300 Buy 2,903 300 Buy 580,800 60,000 Buy 1,362 1,544 Buy 1,362 1,44 Buy 1,365 141 Buy 1,366 145 Buy 1,366 1,475 Buy 2,178 2.25 Buy 2,178 2.25 Buy 2,176 1,427 Buy 2,178 7.26 Buy 2,178 301 Buy 2,914 301 Buy 2,316 46 Buy 2,347 46 Buy 4,250 439 Buy 4,250 439 Buy 4,250 436 Buy 2,547 40 Buy 2,547 40 Buy 2,552 2,74 Buy 2,552 2,74 Buy 2,552 2,74	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	7,744	800	Ordinary
Buy 33,853 3,499 Buy 13,680 60,000 Buy 19,290 1,760 Buy 17,037 1,760 Buy 13,398 2,005 Buy 13,598 2,005 Buy 14,104 1,457 Buy 2,178 1,25 Buy 2,178 1,22 Buy 2,316 1,422 Buy 2,316 1,42 Buy 2,314 301 Buy 2,314 301 Buy 2,314 301 Buy 2,314 130 Buy 2,314 130 Buy 445 46 Buy 1,384 148 Buy 4,250 439 Buy 4,250 439 Buy 4,250 439 Buy 2,594 2,652 Buy 2,594 2,652 Buy 2,652 2,74	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buv	2,903	300	Ordinary
Buy 580,800 60,000 Buy 149,421 15,444 Buy 17,033 1,060 Buy 1,365 141 Buy 7,55 78 Buy 13,75 75 Buy 1,176 2,25 Buy 1,176 7,25 Buy 7,376 1,427 Buy 7,376 1,427 Buy 2,914 301 Buy 2,914 301 Buy 2,914 46 Buy 2,914 14 Buy 2,914 13 Buy 2,914 46 Buy 2,914 13 Buy 2,391 2,47 Buy 4,250 43 Buy 4,250 43 Buy 4,250 43 Buy 2,594 268 Buy 2,594 268 Buy 2,594 268 <t< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>Buy</td><td>33.853</td><td>3 499</td><td>Ordinary</td></t<>	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	33.853	3 499	Ordinary
Buy 149,421 15,444 Buy 17,037 1,760 Buy 19,398 2,005 Buy 1,365 1,41 Buy 1,365 1,41 Buy 1,41,04 1,425 Buy 1,376 1,225 Buy 2,178 2,25 Buy 2,914 301 Buy 2,914 301 Buy 2,314 46 Buy 2,391 2,47 Buy 2,391 2,47 Buy 1,384 1,38 Buy 1,384 1,38 Buy 4,250 45 Buy 2,594 2,68 Buy 2,652 2,74 Buy 2,652 2,74 <td>UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd</td> <td>Birv</td> <td>580.800</td> <td>60,000</td> <td>Ordinary</td>	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Birv	580.800	60,000	Ordinary
Buy 17,037 1,060 Buy 19,396 2,005 Buy 13,398 2,005 Buy 755 78 Buy 14,104 1,457 Buy 13,786 1,225 Buy 1,376 1,62 Buy 2,914 301 Buy 5,814 301 Buy 5,814 301 Buy 5,814 301 Buy 5,814 101 Buy 2,814 101 Buy 445 46 Buy 445 46 Buy 1,384 1,18 Buy 4,250 439 Buy 4,250 439 Buy 4,250 439 Buy 4,250 2,594 Buy 2,594 2,652 Buy 2,652 2,74 Buy 2,652 2,74 Buy 2,594 2,84	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Purk	140,421	15 444	Ordinary
Buy 17,037 1,005 Buy 19,398 2,005 Buy 1,366 141 Buy 1,366 145 Buy 2,178 2.55 Buy 1,376 1,422 Buy 7,376 7.25 Buy 7,376 7.22 Buy 2,914 301 Buy 2,914 301 Buy 2,914 46 Buy 2,914 46 Buy 2,391 247 Buy 1,384 113 Buy 4,250 439 Buy 4,250 439 Buy 4,250 439 Buy 2,594 2,685 Buy 2,652 2,74 Buy 2,652 2,74 Buy 2,652 2,74 Buy 2,652 2,74	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	ng	143,421	13,444	Cidinaly
Buy 1,358 1,005 Buy 1,358 1,005 Buy 1,355 78 Buy 2,178 2,25 Buy 2,176 1,422 Buy 1,376 1,422 Buy 2,314 301 Buy 2,314 301 Buy 445 46 Buy 2,391 247 Buy 1,384 143 Buy 1,384 143 Buy 4,250 439 Buy 4,250 459 Buy 4,250 45 Buy 4,250 45 Buy 4,250 45 Buy 2,594 2,68 Buy 2,594 2,68 Buy 2,652 2,74 Buy 2,594 2,74 Buy 2,594 2,74 Buy 2,594 2,74 Buy 2,594 2,74 <t< td=""><td>UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd</td><td>Buy</td><td>17,037</td><td>1,/60</td><td>Ordinary</td></t<>	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	17,037	1,/60	Ordinary
Buy 7.565 141 Buy 7.55 7.84 Buy 14,104 1,457 Buy 13,765 1,457 Buy 13,765 1,452 Buy 7,376 7.62 Buy 7,376 7.67 Buy 7,570 4.39 Buy 7,565 7.64 Buy 7,565 7.64 Buy 7,565 7.64 Buy 7,565 7.74 Buy 7,565 7.	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	19,398	2,005	Ordinary
Buy 755 78 Buy 14,104 1,475 Buy 2,178 2.25 Buy 13,765 1,422 Buy 7,376 1,422 Buy 7,376 1,422 Buy 2,914 301 Buy 2,914 60 Buy 2,914 40 Buy 2,391 2,47 Buy 1,38 1,38 Buy 1,38 1,38 Buy 4,250 4,99 Buy 4,250 4,99 Buy 2,594 2,68 Buy 2,652 2,74	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	1,365	141	Ordinary
Buy 1,470 1,457 Buy 2,178 1,422 Buy 1,376 1,225 Buy 1,376 762 Buy 2,314 301 Buy 581 60 Buy 445 46 Buy 1,391 2,47 Buy 1,384 1,43 Buy 1,384 1,48 Buy 4,550 439 Buy 4,560 45 Buy 2,594 2,684 Buy 2,652 2,68 Buy 2,652 2,74 Buy 2,652 2,74 Buy 2,652 2,74 Buy 2,652 2,74	UBS Securities Australia ttd UBS Securities Australia ttd UBS Securities Australia ttd UBS Securities Australia ttd UBS Securities Australia ttd UBS Securities Australia ttd UBS Securities Australia ttd UBS Securities Australia ttd	Buy	755	78	Ordinary
Buy 2,178 225 Buy 13,765 1,422 Buy 7,376 762 Buy 2,914 301 Buy 2,914 46 Buy 445 46 Buy 2,391 247 Buy 1,384 113 Buy 1,386 138 Buy 4,250 439 Buy 4,250 439 Buy 2,594 2,685 Buy 2,652 2,74	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	14,104	1,457	Ordinary
Buy 13,765 1,422 Buy 7,376 762 Buy 2,914 701 Buy 2,914 60 Buy 2,914 60 Buy 2,391 247 Buy 1,394 143 Buy 1,384 143 Buy 4,250 439 Buy 4,250 45 Buy 2,594 268 Buy 2,594 268 Buy 2,652 274 Buy 3,87 404	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	2,178	225	Ordinary
Buy 7,376 762 Buy 2,914 301 Buy 581 60 Buy 445 46 Buy 2,391 247 Buy 1,384 143 Buy 1,326 138 Buy 4,250 49 Buy 4,250 49 Buy 2,594 2,68 Buy 2,594 2,68 Buy 3,7 40 Buy 2,652 274 Buy 3,7 40	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	13,765	1,422	Ordinary
Buy 2,914 301 Buy 2,914 301 Buy 445 465 Buy 2,391 247 Buy 1,394 1,38 Buy 1,396 1,38 Buy 4,250 4,39 Buy	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	7.376	762	Ordinary
Buy 581 60 Buy 581 60 Buy 445 46 Buy 2,391 247 Buy 1,384 1,43 Buy 1,386 1,38 Buy 4,250 439 Buy 2,594 2,68 Buy 2,562 274 Buy 327 40	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	2 914	301	Ordinary
Buy A45 A6 Buy 2,391 247 Buy 1,384 143 Buy 1,386 138 Buy 4,250 439 Buy 2,594 268 Buy 2,562 274 Buy 3,37 A0	UBS Securities Australia Ltd	600	183	100	Ordinary
Buy 445 46 Buy 2,391 247 Buy 1,394 143 Buy 1,336 138 Buy 4,250 439 Buy 4,250 45 Buy 2,594 268 Buy 2,594 268 Buy 387 204	UBS Securities Australia Ltd	(na	ō.	3 :	Cluring
Buy 247 Buy 10 1 Buy 1,384 1,43 Buy 1,386 1,38 Buy 4,250 4,39 Buy 2,594 2,68 Buy 2,594 2,68 Buy 2,652 274 Buy 347 400	It IDC Committee Ametralia to	Buy	445	46	Ordinary
Blay 10 1 Buy 1,384 143 Buy 1,386 138 Buy 4,250 439 Buy 4,250 436 Buy 2,544 268 Buy 2,544 268 Buy 387 20	ODS Securities Australia הגם	Buy	2,391	247	Ordinary
Buy 1,384 143 Buy 1,336 138 Buy 4,250 439 Buy 4,26 45 Buy 2,594 268 Buy 2,594 268 Buy 387 201	UBS Securities Australia Ltd	Buy	10	1	Ordinary
8uy 1,336 138 8uy 4,250 4.9 8uy 4,250 4.9 8uy 2,594 268 8uy 2,594 268 8uy 327 274	UBS Securities Australia Ltd	Buy	1,384	143	Ordinary
Buy 4.250 439 Buy 436 45 Buy 2.594 268 Buy 2.682 274 Buy 387 400	UBS Securities Australia Ltd	Buy	1,336	138	Ordinary
6.0y 456 Buy 2,562 274 Buy 2,562 274 Buy 337 40	I IRC Cocumities Australia I to	Bary	4 250	057	Ordinary
Buy 2,594 268 Buy 2,594 268 Buy 2,652 274 Buy 387 40	ODS Securities Australia Ltd	, and	25,4	יין אין	Coming
Buy 2.594 268 Buy 2.652 274 Buy 387 40	UBS Securities Australia Ltd	Buy	436	45	Ordinary
Buy 2,652 274	UBS Securities Australia Ltd	Buy	2,594	268	Ordinary
287 40	UBS Securities Australia Ltd	Buy	2,652	274	Ordinary
	De Continue Assets	, ,	207	OV	Ordina
8dy +0	UBS Securities Australia Ltd	Buy	38/	40	Ordinary

(85) Securities Australia Ltd Buy 28,943 2,990 (85) Securities Australia Ltd Buy 28,943 2,990 (85) Securities Australia Ltd Buy 3,078 3,788 2,88 (85) Securities Australia Ltd Buy 3,078 3,198 3,78 (85) Securities Australia Ltd Buy 3,428 3,51 4,68 (85) Securities Australia Ltd Buy 3,428 3,52 3,51 (85) Securities Australia Ltd Buy 3,428 3,52 3,51 (85) Securities Australia Ltd Buy 3,428 3,52 3,61 (85) Securities Australia Ltd Buy 3,428 3,52 3,61 4,62 (85) Securities Australia Ltd Buy 3,428 3,52 3,61 4,62 3,61 4,62 3,62 3,61 4,62 3,62 3,61 4,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62 3,62	10 Lat 1.4	IDC Committee Assessed to		7 82		Ciginary
LIST Scorring Autorials List By/L 2,788 2,788 LIST Scorring Autorials List Buy 2,788 3,614 LIST Scorring Autorials List Buy 19,531 20,738 LIST Scorring Autorials List Buy 1,227 3,614 LIST Scorring Autorials List Buy 1,227 3,521 LIST Scorring Autorials List Buy 3,238 3,511 LIST Scorring Autorials List Buy 3,238 3,521 LIST Scorring Autorials List Buy 3,238 3,521 LIST Scorring Autorials List Buy 3,522 3,60 LIST Scorring Autorials List Buy 3,60 3,00 LIST Scorring Autorials List Buy 3,738 3,00 LIST Scorring Autorials List Buy 3,738 3,00 LIST Scorring Autorials List Buy 2,738 3,00 LIST Scorring Autorials List Buy 2,738 3,00 LIST Scorring Autorials List Buy 2,738 3,00 LIST Scorring Autorials List Bu	8-Jul-14	UBS Securities Australia (1d	Buy	28 943	2 990	Ordinan
USS Securites Australia Ltd Buy 19,258 20,78 31,18 USS Securites Australia Ltd Buy 19,259 31,21 USS Securites Australia Ltd Buy 1,227 31,21 USS Securites Australia Ltd Buy 3,271 3,271 271 USS Securites Australia Ltd Buy 3,271 3,271 3,271 USS Securites Australia Ltd Buy 3,271 3,271 3,271 USS Securites Australia Ltd Buy 3,271 3,271 3,271 USS Securites Australia Ltd Buy 3,271 3,271 3,271 3,271 USS Securites Australia Ltd Buy 3,271 3,	8-Jul-14	UBS Securities Australia I td	Buy	2,022	288	Ordinary
USS Scorrifies Australia Ltd Buy 1,95,581 3,01,01 USS Scorrifies Australia Ltd Buy 1,92,591 3,01,01 USS Scorrifies Australia Ltd Buy 1,227 3,22 96 USS Scorrifies Australia Ltd Buy 1,237 3,251 1,88 USS Scorrifies Australia Ltd Buy 3,236 3,20 3,20 USS Scorrifies Australia Ltd Buy 3,236 3,20 3,20 USS Scorrifies Australia Ltd Buy 3,20 3,20 3,00 USS Scorrifies Australia Ltd Buy 3,00 3,00 3,00 3,00 USS Scorrifies Australia Ltd Buy 3,00 3,00 3,00 3,00 3,00 3,00 3,00 3,00 3,00 3,00 3,00 3,00 3,00 3,0	8-14-14	I IBS Securities Australia I to	Birk	3 0 20	210	Ordinary
URS Scorrifes Australia Ltd Buy 3,49,891 1,09,31 URS Scorrifes Australia Ltd Buy 3,49,81 1,04 URS Scorrifes Australia Ltd Buy 1,27 3,11 URS Scorrifes Australia Ltd Buy 3,20 312 URS Scorrifes Australia Ltd Buy 3,20 312 URS Scorrifes Australia Ltd Buy 3,20 312 URS Scorrifes Australia Ltd Buy 3,20 326 URS Scorrifes Australia Ltd Buy 3,20 326 URS Scorrifes Australia Ltd Buy 3,156 3,20 URS Scorrifes Australia Ltd Buy 3,156	1111	Upo con miss Australia utu	had	9/0'5	318	Ordinary
UBS Scorrifes Australia Ltd Buy 34,984 3,614 UBS Scorrifes Australia Ltd Buy 1,227 9,6 UBS Scorrifes Australia Ltd Buy 1,227 1,18 UBS Scorrifes Australia Ltd Buy 3,271 2,51 UBS Scorrifes Australia Ltd Buy 3,272 2,6 UBS Scorrifes Australia Ltd Buy 3,128 2,6 UBS Scorrifes Australia Ltd Buy 3,128 2,8 UBS Scorrifes Australia Ltd Buy 3,128 2,8 UBS Scorrifes Australia Ltd Buy 3,156 2,9 UBS Scorrifes Australia Ltd Buy 3,156	1 1 1	OBS Securities Adstraila Ltd	Buy	186,081	20,308	Ordinary
USS Securities Australia Ltd Buy 1,297 134 USS Securities Australia Ltd Buy 1,297 134 USS Securities Australia Ltd Buy 3,200 310 USS Securities Australia Ltd Buy 3,200 310 USS Securities Australia Ltd Buy 8,230 326 USS Securities Australia Ltd Buy 8,156 326 USS Securities Australia Ltd Buy 2,178 326 USS Securities Australia Ltd Buy 2,726 326 USS Securities Australia Ltd Buy 2,726 326 USS Securities Australia Ltd Buy 2,726<	9-Jul-14	Ubs Securities Australia Ltd	Buy	34,984	3,614	Ordinary
UBS Securities Australia Ltd Buy 917 1142 1162 UBS Securities Australia Ltd Buy 917 917 118 UBS Securities Australia Ltd Buy 822 24 UBS Securities Australia Ltd Buy 822 24 UBS Securities Australia Ltd Buy 918 222 24 UBS Securities Australia Ltd Buy 218 256 26 UBS Securities Australia Ltd Buy 2113 25 26 UBS Securities Australia Ltd Buy 2112 26 26 UBS Securities Australia Ltd Buy 2113 26 26 UBS Securities Australia Ltd Buy 2112 26 26 UBS Securities Australia Ltd Buy 2128 26 26	8-JUI-14	UBS Securities Australia Ltd	Buy	1,297	134	Ordinary
UBS Securities Autorials Ltd Blyy 31,142 11,8 UBS Securities Autorials Ltd Blyy 34,02 31,0 UBS Securities Autorials Ltd Blyy 3,020 31,0 UBS Securities Autorials Ltd Blyy 3,020 37,0 UBS Securities Autorials Ltd Blyy 3,172 2,8 UBS Securities Autorials Ltd Blyy 3,172 2,80 UBS Securities Autorials Ltd Blyy 3,173 3,20 UBS Securities Autorials Ltd Blyy 3,173 3,20 UBS Securities Autorials Ltd Blyy 3,173 3,20 UBS Securities Autorials Ltd Blyy 3,178 3,18 UBS Securities Autorials Ltd Blyy 3,18 3,18 UBS Securities Autorials Ltd Blyy 3,18 3,18 UBS Securities Autorials Ltd <th< td=""><td>8-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>929</td><td>96</td><td>Ordinary</td></th<>	8-Jul-14	UBS Securities Australia Ltd	Buy	929	96	Ordinary
UBS Securities Autoraties Ltd By part and a 34,74 3,51 3,51 UBS Securities Autoraties Ltd Buy 622 88 UBS Securities Autoraties Ltd Buy 622 82 UBS Securities Autoraties Ltd Buy 232 23 UBS Securities Autoraties Ltd Buy 3,156 2,56 UBS Securities Autoraties Ltd Buy 3,156 2,58 UBS Securities Autoraties Ltd Buy 2,177 2,58 UBS Securities Autoraties Ltd Buy 2,177 2,58 UBS Securities Autoraties Ltd Buy 2,177 2,58 UBS Securities Autoraties Ltd Buy 2,172 2,58 UBS Securities Autoraties Ltd Buy 4,152 2,58 UBS Securities Autoraties Ltd Buy 4,152 1,58 UBS Securities Autoraties Ltd Buy 4,152 1,58 UBS Securities Autoraties Ltd Buy 1,152 1,175 UBS Securities Autoraties Ltd Buy 2,172 2,18 UBS Securities Autoraties Ltd	8-Jul-14	UBS Securities Australia Ltd	Buy	1,142	118	Ordinary
UBS Securities Autoraties Ltd By purple 3,000 31.7 UBS Securities Autoraties Ltd Buy 6,228 65.1 UBS Securities Autoraties Ltd Buy 2,152 2.6 UBS Securities Autoraties Ltd Buy 3,156 2.6 UBS Securities Autoraties Ltd Buy 3,177 2,600 UBS Securities Autoraties Ltd Buy 3,177 2,801 UBS Securities Autoraties Ltd Buy 3,177 2,801 UBS Securities Autoraties Ltd Buy 3,177 2,801 UBS Securities Autoraties Ltd Buy 3,177 3,201 UBS Securities Autoraties Ltd Buy 3,178 3,178 UBS Securities Autoraties Ltd Buy 3,178 3,18 UBS Securities Autoraties Ltd Buy 3,128 3,18 UBS Securities Autoraties Ltd<	8-Jul-14	UBS Securities Australia Ltd	Buy	34,374	3,551	Ordinary
UBS Securities Australia Ltd Buy 68.2 88 UBS Securities Australia Ltd Buy 2.22 2.4 UBS Securities Australia Ltd Buy 3.15G 2.6 UBS Securities Australia Ltd Buy 3.15G 2.6 UBS Securities Australia Ltd Buy 3.15G 2.6 UBS Securities Australia Ltd Buy 2.7 2.8 UBS Securities Australia Ltd Buy 4.62 2.0 UBS Securities Australia Ltd Buy 4.7 2.0 UBS Securities Australia Ltd Buy 5.02 2.0 UBS Securities Australia Ltd Buy 5.02 2.	8-Jul-14	UBS Securities Australia Ltd	Buy	3,020	312	Ordinary
UBS Securities Australia Ltd Bay 8.238 85.1 1 UBS Securities Australia Ltd Buy 3,156 3.26 1 UBS Securities Australia Ltd Buy 3,172 2.80 1 UBS Securities Australia Ltd Buy 3,173 3,00 1 UBS Securities Australia Ltd Buy 3,173 1,75 1 UBS Securities Australia Ltd Buy 1,026 1,00 1 UBS Securities Australia Ltd Buy 2,04 3,00 1 UBS Securities Australia Ltd Buy 2,04 3,00 1 UBS Securities Australia Ltd Buy 4,172 4,10 1 UBS Securities Australia Ltd Buy 2,04 3,00 1 UBS Securities Australia Ltd Buy 2,178 3,00 1 UBS Securities Australia Ltd Buy 2,178 3,00 1 UBS Securities Australia Ltd <td>8-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>852</td> <td>88</td> <td>Ordinary</td>	8-Jul-14	UBS Securities Australia Ltd	Buv	852	88	Ordinary
UBS Securities Australia Ltd BM 3.75 2.4 UBS Securities Australia Ltd BMy 3,156 3.26 UBS Securities Australia Ltd BMy 3,156 3.26 UBS Securities Australia Ltd BMy 2,777 2,807 UBS Securities Australia Ltd BMy 2,778 3.09 UBS Securities Australia Ltd BMy 2,778 3.00 UBS Securities Australia Ltd BMy 2,778 3.00 UBS Securities Australia Ltd BMy 2,778 3.00 UBS Securities Australia Ltd BMy 1,755 1,75 UBS Securities Australia Ltd BMy 1,756 1,56 UBS Securities Australia Ltd BMy 1,756 1,57 UBS Securities Australia Ltd BMy 4,172 4,10 UBS Securities Australia Ltd BMy 1,256 4,60 UBS Securities Australia Ltd BMy 1,756 1,76 UBS Securities Australia Ltd BMy 1,756 1,76 UBS Securities Australia Ltd BMy	8-111-14	LIBS Securities Australia Ltd	Bix	8 238	PE-1	Ording
(BS Securities Australia Ltd Buy 3,152 3,26 (BS Securities Australia Ltd Buy 3,156 3,26 (BS Securities Australia Ltd Buy 3,156 3,26 (BS Securities Australia Ltd Buy 3,172 3,69 (BS Securities Australia Ltd Buy 3,173 3,69 (BS Securities Australia Ltd Buy 3,173 3,60 (BS Securities Australia Ltd Buy 3,173 3,60 (BS Securities Australia Ltd Buy 3,173 1,75 (BS Securities Australia Ltd Buy 3,162 3,50 (BS Securities Australia Ltd Buy 3,162 3,50 (BS Securities Australia Ltd Buy 3,162 3,50 (BS Securities Australia Ltd Buy 4,162 4,30 (BS Securities Australia Ltd Buy 4,162 4,30 (BS Securities Australia Ltd Buy 3,156 3,26 (BS Securities Australia Ltd Buy 3,156 3,26 (BS Securities Australia Ltd Buy	R-111-14	IIBS Securities Australia Itd	600	560	3	Granary
UBS Securities Australia Ltd BMy 4,810 520 UBS Securities Australia Ltd Bluy 4,810 3,156 3,26 UBS Securities Australia Ltd Bluy 2,715 3,66 3,67 <		The Commerce Account to 1911	fog 4	767	47	Comary
UBS Securities Australia Ltd Bluy 3,1840 500 UBS Securities Australia Ltd Buy 3,1717 2,807 326 UBS Securities Australia Ltd Buy 3,1717 2,807 326 UBS Securities Australia Ltd Buy 3,1717 2,807 32,604 350 UBS Securities Australia Ltd Buy 3,172 2,904 350 300	17.0	Obs securities Australia Ltd	Buy	3,156	976	Ordinary
UBS Securities Australia Ltd Buy 3156 326 UBS Securities Australia Ltd Buy 27,172 2,80 UBS Securities Australia Ltd Buy 27,04 300 UBS Securities Australia Ltd Buy 2,204 300 UBS Securities Australia Ltd Buy 2,204 300 UBS Securities Australia Ltd Buy 2,204 300 UBS Securities Australia Ltd Buy 1,175 430 UBS Securities Australia Ltd Buy 1,175 1,175 UBS Securities Australia Ltd Buy 1,1	9-701-14	UBS Securities Australia Ltd	Buy	4,840	200	Ordinary
UBS Securities Australia Ltd Buy 57,172 2,807 UBS Securities Australia Ltd Buy 57,172 2,807 UBS Securities Australia Ltd Buy 5,208 5,381 UBS Securities Australia Ltd Buy 7,204 300 UBS Securities Australia Ltd Buy 1,775 1,775 UBS Securities Australia Ltd Buy 7,938 820 UBS Securities Australia Ltd Buy 7,938 820 UBS Securities Australia Ltd Buy 7,172 431 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 6,689 509 UBS Securities Australia Ltd Buy 5,022 506 UBS Securities Australia Ltd Buy 5,028 5,024 510 UBS Securities Australia Ltd Buy 5,028 5,024 510 UBS Securities Australia Ltd Buy 5,028 5,024 510	8-Jul-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
UBS Securities Australia Ltd Buy 9,157 946 UBS Securities Australia Ltd Buy 2,048 3,094 UBS Securities Australia Ltd Buy 2,048 3,00 UBS Securities Australia Ltd Buy 2,048 3,00 UBS Securities Australia Ltd Buy 1,1378 1,157 UBS Securities Australia Ltd Buy 2,478 1,256 UBS Securities Australia Ltd Buy 2,678 1,056 UBS Securities Australia Ltd Buy 2,678 2,60 UBS Securities Australia Ltd Buy 2,788 5,00 UBS Securities Australia Ltd Buy 5,02 5,00 UBS Securities Australia Ltd Buy 5,02 5,00 UBS Securities Australia Ltd Buy 5,024 5,00 UBS Securities Australia Ltd Buy 5,024 5,00 UBS Securities Australia Ltd Buy 7,125 1,03 UBS Securities Australia Ltd Buy 7,125 1,03 UBS Securities Australia Ltd Buy	8-Jul-14	UBS Securities Australia Ltd	Buy	27,172	2,807	Ordinary
UBS Securities Australia Ltd Bly 52,088 5,381 UBS Securities Australia Ltd Bly 1,737 1,736 UBS Securities Australia Ltd Bly 1,1374 1,737 UBS Securities Australia Ltd Bly 1,1374 1,737 UBS Securities Australia Ltd Bly 1,028 2,048 UBS Securities Australia Ltd Bly 1,028 2,029 UBS Securities Australia Ltd Bly 1,028 2,020 UBS Securities Australia Ltd Bly 1,028 2,020 UBS Securities Australia Ltd Bly 1,038 2,038 UBS Securities Australia Ltd Bly 1,038 2,048 UBS Securities Australia Ltd Bly 4,152 1,01 UBS Securities Australia Ltd Bly 1,725 1,739 UBS Securities Australia Ltd Bly 1,725 1,739 UBS Securities Australia Ltd Bly 1,725 1,730 UBS Securities Australia Ltd Bly 1,725 1,725 UBS Securities Australia Ltd	8-Jul-14	UBS Securities Australia Ltd	Suc.	9.157	946	Ordinary
UBS Securities Australia Ltd Bay 2,564 500 UBS Securities Australia Ltd Bay 1,175 4,162 1,37 UBS Securities Australia Ltd Buy 1,175 4,10 UBS Securities Australia Ltd Buy 7,938 820 UBS Securities Australia Ltd Buy 6,688 506 UBS Securities Australia Ltd Buy 6,688 506 UBS Securities Australia Ltd Buy 5,024 4,172 44 UBS Securities Australia Ltd Buy 5,024 4,172 44 UBS Securities Australia Ltd Buy 7,123 1,08 506 UBS Securities Australia Ltd Buy 7,123 1,08 300 UBS Securities Australia Ltd Buy 7,123 1,08 300 UBS Securities Australia Ltd Buy 7	8-Jul-14	UBS Securities Australia Ltd	Site	52 088	5 381	Ordinary
USS Securities Australia Ltd Buy 4,125 4,30 UBS Securities Australia Ltd Buy 1,026 1,05 UBS Securities Australia Ltd Buy 7,938 80 UBS Securities Australia Ltd Buy 4,162 430 UBS Securities Australia Ltd Buy 4,162 430 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 7,183 1,783 UBS Securities Australia Ltd Buy 7,183 1,783 UBS Securities Australia Ltd Buy 7,183 1,783 UBS Securities Australia Ltd Buy 5,278 2,78 UBS Securities Australia Ltd Buy	8-11:1-14	I BS Securities Australia Ltd	2 6	7000	0000	in in in
UBS Securities Australia Ltd Buy 1,1374 1,75 UBS Securities Australia Ltd Buy 1,237 1,75 UBS Securities Australia Ltd Buy 2,478 1,56 UBS Securities Australia Ltd Buy 7,938 820 UBS Securities Australia Ltd Buy 7,938 820 UBS Securities Australia Ltd Buy 7,938 820 UBS Securities Australia Ltd Buy 6,89 5,99 97 UBS Securities Australia Ltd Buy 4,189 5,06 101 UBS Securities Australia Ltd Buy 4,172 431 401 UBS Securities Australia Ltd Buy 5,286 5,66 5,286 56 UBS Securities Australia Ltd Buy 5,228 5,68 5,99 95 UBS Securities Australia Ltd Buy 5,228 5,68 5,98 5,178 1,178 UBS Securities Australia Ltd Buy 5,228 5,28 5,178 1,178 UBS Securities Australia Ltd Buy		I loc contribute Australia Ltd	soo d	2,304	300	Ordinary
UBS Securities Australia Ltd Buy 1,1374 1,775 UBS Securities Australia Ltd Buy 2,478 2,68 UBS Securities Australia Ltd Buy 7,938 1,05 UBS Securities Australia Ltd Buy 7,938 8.20 UBS Securities Australia Ltd Buy 5,092 5.26 UBS Securities Australia Ltd Buy 4,182 2.00 UBS Securities Australia Ltd Buy 4,182 2.00 UBS Securities Australia Ltd Buy 4,182 2.00 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 5,286 566 UBS Securities Australia Ltd Buy 7,163 70 UBS Securities Australia Ltd Buy 7,163 70 UBS Securities Australia Ltd Buy 7,184 287 UBS Securities Australia Ltd Buy 7,184 38 UBS Securities Australia Ltd Buy <th< td=""><td>#1 -Inf-0</td><td>Obs Securities Australia Ltd</td><td>and</td><td>4,162</td><td>430</td><td>Ordinary</td></th<>	#1 -Inf-0	Obs Securities Australia Ltd	and	4,162	430	Ordinary
UBS Securities Australia Ltd Bly 2,478 266 UBS Securities Australia Ltd Buy 1,026 106 UBS Securities Australia Ltd Buy 7,538 820 15 UBS Securities Australia Ltd Buy 6,689 506 202 505 UBS Securities Australia Ltd Buy 4,162 203 506 UBS Securities Australia Ltd Buy 4,888 506 207 UBS Securities Australia Ltd Buy 4,172 431 406 UBS Securities Australia Ltd Buy 4,172 431 406 UBS Securities Australia Ltd Buy 5,285 546 507 UBS Securities Australia Ltd Buy 5,285 546 508 506 UBS Securities Australia Ltd Buy 5,235 546 507 178 507 508 UBS Securities Australia Ltd Buy 5,236 546 508 508 508 508 508 508 508 508 508	8-Jul-14	UBS Securities Australia Ltd	Buy	11,374	1,175	Ordinary
UBS Securities Australia Ltd Buy 1,026 106 UBS Securities Australia Ltd Buy 7,938 106 UBS Securities Australia Ltd Buy 7,938 256 UBS Securities Australia Ltd Buy 4,659 256 UBS Securities Australia Ltd Buy 4,899 200 UBS Securities Australia Ltd Buy 4,899 26 UBS Securities Australia Ltd Buy 4,899 26 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 7,163 1,783 UBS Securities Australia Ltd Buy 7,162 2,72 UBS Securities Australia Ltd Buy 7,723 1,78 UBS Securities Australia Ltd Buy <t< td=""><td>8-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>2,478</td><td>526</td><td>Ordinary</td></t<>	8-Jul-14	UBS Securities Australia Ltd	Buy	2,478	526	Ordinary
UBS Securities Australia Ltd Buy 7,938 B20 UBS Securities Australia Ltd Buy 1,45 15 UBS Securities Australia Ltd Buy 4,162 200 UBS Securities Australia Ltd Buy 4,898 509 UBS Securities Australia Ltd Buy 4,898 509 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 4,172 43 UBS Securities Australia Ltd Buy 1,723 1,783 UBS Securities Australia Ltd Buy 1,723 1,783 UBS Securities Australia Ltd Buy 1,723 1,784 UBS Securities Australia Ltd Buy 5,778 1,784 UBS Securities Australia Ltd Buy 1,723 1,784 UBS Securities Australia Ltd Buy <td< td=""><td>3-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1,026</td><td>106</td><td>Ordinary</td></td<>	3-Jul-14	UBS Securities Australia Ltd	Buy	1,026	106	Ordinary
UBS Securities Australia Ltd Buy 5,092 5,09 UBS Securities Australia Ltd Buy 4,162 400 UBS Securities Australia Ltd Buy 4,182 400 UBS Securities Australia Ltd Buy 4,898 50 UBS Securities Australia Ltd Buy 4,898 50 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 5,285 5.46 UBS Securities Australia Ltd Buy 5,285 5.46 UBS Securities Australia Ltd Buy 5,788 5.46 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 5,778 25 UBS Securities Australia Ltd Buy 5,724 1,09 UBS Securities Australia Ltd Buy 5,724 1,09 UBS Securities Australia Ltd Buy 5,748 2,724 UBS Securities Australia Ltd Buy <td< td=""><td>8-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buv</td><td>7.938</td><td>820</td><td>Ordinary</td></td<>	8-Jul-14	UBS Securities Australia Ltd	Buv	7.938	820	Ordinary
UBS Securities Australia Ltd Buy 5,092 5,56 UBS Securities Australia Ltd Buy 1,782 430 UBS Securities Australia Ltd Buy 1,782 200 UBS Securities Australia Ltd Buy 4,893 506 UBS Securities Australia Ltd Buy 4,772 431 UBS Securities Australia Ltd Buy 5,285 5,66 UBS Securities Australia Ltd Buy 4,772 431 UBS Securities Australia Ltd Buy 5,786 5,46 UBS Securities Australia Ltd Buy 5,786 5,86 UBS Securities Australia Ltd Buy 5,786 3,87 UBS Securities Australia Ltd Buy 5,024 5,97 UBS Securities Australia Ltd Buy	8-Jul-14	UBS Securities Australia Itd	Riv	145	12	Ordinan
UBS Securities Australia Ltd Buy 4,162 430 UBS Securities Australia Ltd Buy 4,162 430 UBS Securities Australia Ltd Buy 4,889 506 UBS Securities Australia Ltd Buy 4,889 506 UBS Securities Australia Ltd Buy 4,772 431 UBS Securities Australia Ltd Buy 4,772 431 UBS Securities Australia Ltd Buy 7,163 546 UBS Securities Australia Ltd Buy 7,163 1,783 UBS Securities Australia Ltd Buy 7,163 1,783 UBS Securities Australia Ltd Buy 7,163 1,783 UBS Securities Australia Ltd Buy 7,763 1,87 UBS Securities Australia Ltd Buy 2,778 2,87 UBS Securities Australia Ltd Buy 2,724 49 UBS Securities Australia Ltd Buy 2,724 26 UBS Securities Australia Ltd Buy 2,724 26 UBS Securities Australia Ltd Buy 2	2. 1. 1. 1. 4	Let Constitute Assessed at the		2007	2 2	, i
UBS Securities Australia Ltd Buy 4,162 400 UBS Securities Australia Ltd Buy 6,689 691 UBS Securities Australia Ltd Buy 4,789 506 UBS Securities Australia Ltd Buy 939 971 UBS Securities Australia Ltd Buy 4,772 431 UBS Securities Australia Ltd Buy 4,772 431 UBS Securities Australia Ltd Buy 4,772 431 UBS Securities Australia Ltd Buy 7,763 5,285 5,46 UBS Securities Australia Ltd Buy 7,763 1,783 1,783 1,783 UBS Securities Australia Ltd Buy 1,725 1,783 1,784 1,785 1,783 1,784 1,784 1,785 1,784 1,785 1,784 1,784 1,785 1,784 1,785 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,784 1,7		COO December Againatia Lid	(ng	2,032	270	Ordinary
UBS Securities Australia Ltd Buy 6,898 601 UBS Securities Australia Ltd Buy 4,898 506 UBS Securities Australia Ltd Buy 973 101 UBS Securities Australia Ltd Buy 9,78 101 UBS Securities Australia Ltd Buy 5,285 546 UBS Securities Australia Ltd Buy 7,163 7,40 UBS Securities Australia Ltd Buy 7,753 1,783 UBS Securities Australia Ltd Buy 3,156 366 UBS Securities Australia Ltd Buy 3,156 378 UBS Securities Australia Ltd Buy 3,736 387 UBS Securities Australia Ltd Buy 3,736 387 UBS Securities Australia Ltd Buy 3,736 386 UBS Securities Australia Ltd Buy 3,736 386 UBS Securities Australia Ltd Buy 5,737 36 UBS Securities Australia Ltd Buy 3,736 36 UBS Securities Australia Ltd Buy 2,737 <td>s-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>4,162</td> <td>430</td> <td>Ordinary</td>	s-Jul-14	UBS Securities Australia Ltd	Buy	4,162	430	Ordinary
UBS Securities Australia Ltd Buy 6,689 691 UBS Securities Australia Ltd Buy 939 506 UBS Securities Australia Ltd Buy 978 101 UBS Securities Australia Ltd Buy 978 101 UBS Securities Australia Ltd Buy 5,285 546 UBS Securities Australia Ltd Buy 7,163 740 UBS Securities Australia Ltd Buy 7,163 740 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 5,778 2,774 UBS Securities Australia Ltd Buy 5,74 1,054 UBS Securities Australia Ltd Buy 2,7	3-Jul-14	UBS Securities Australia Ltd	Buy	1,936	200	Ordinary
UBS Securities Australia Ltd Buy 4,898 506 UBS Securities Australia Ltd Buy 978 101 UBS Securities Australia Ltd Buy 978 101 UBS Securities Australia Ltd Buy 5,285 546 UBS Securities Australia Ltd Buy 7,163 740 UBS Securities Australia Ltd Buy 1,7259 1,783 UBS Securities Australia Ltd Buy 1,7259 1,783 UBS Securities Australia Ltd Buy 1,725 150 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 1,723 1,089 UBS Securities Australia Ltd Buy 1,723 1,089 UBS Securities Australia Ltd Buy 2,778 2,67 UBS Securities Australia Ltd Buy 2,724 2,69 UBS Securities Australia Ltd Buy 2,724 2,69 UBS Securities Australia Ltd Buy <	3-Jul-14	UBS Securities Australia Ltd	Buv	689.9	691	Ordinary
USS Securities Australia Ltd Buy 978 971 UBS Securities Australia Ltd Buy 978 101 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 5,285 5,46 UBS Securities Australia Ltd Buy 5,285 5,46 UBS Securities Australia Ltd Buy 1,753 1,733 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 2,778 1,78 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 5,724 1,09 UBS Securities Australia Ltd Buy 5,724 1,09 UBS Securities Australia Ltd Buy 2,73 1,05 UBS Securities Australia Ltd Buy 2,74 2,73 UBS Securities Australia Ltd Buy	8-1:1-14	LIBS Securities Australia 11d	Buy	7 808	202	Ordingo
UBS Securities Australia Ltd Buy 978 101 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 4,172 431 UBS Securities Australia Ltd Buy 5,285 546 UBS Securities Australia Ltd Buy 7,163 740 UBS Securities Australia Ltd Buy 4,774 49 UBS Securities Australia Ltd Buy 3,156 35 UBS Securities Australia Ltd Buy 1,723 150 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 1,723 1,089 UBS Securities Australia Ltd Buy 1,723 1,089 UBS Securities Australia Ltd Buy 2,778 26 UBS Securities Australia Ltd Buy 2,722 26 UBS Securities Australia Ltd Buy 2,732 26 UBS Securities Australia Ltd Buy 2,732 <td>11 34</td> <td>Late of the state</td> <td>650</td> <td>000</td> <td>200</td> <td>A</td>	11 34	Late of the state	650	000	200	A
UBS Securities Australia Ltd Bby 4,172 431 UBS Securities Australia Ltd Bby 4,172 431 UBS Securities Australia Ltd Bby 5,285 546 UBS Securities Australia Ltd Bby 7,163 740 UBS Securities Australia Ltd Bby 7,163 740 UBS Securities Australia Ltd Bby 1,723 1,783 UBS Securities Australia Ltd Bby 1,723 1,78 UBS Securities Australia Ltd Bby 2,778 386 UBS Securities Australia Ltd Bby 1,723 1,78 UBS Securities Australia Ltd Bby 2,778 945 UBS Securities Australia Ltd Bby 2,724 1,059 UBS Securities Australia Ltd Bby 2,724 1,059 UBS Securities Australia Ltd Bby 2,724 1,059 UBS Securities Australia Ltd Bby 2,724 2,724 UBS Securities Australia Ltd Bby 2,724 2,724 UBS Securities Australia Ltd Bby	9-JUI-14	Opp Securities Australia Liu	buy	939	76	Ordinary
UBS Securities Australia Ltd Bby 4,172 431 UBS Securities Australia Ltd Buy 5,285 546 UBS Securities Australia Ltd Buy 7,163 740 UBS Securities Australia Ltd Buy 7,163 740 UBS Securities Australia Ltd Buy 7,756 3,756 3,756 UBS Securities Australia Ltd Buy 7,778 1,87 1,90 UBS Securities Australia Ltd Buy 2,778 2,87 1,78 UBS Securities Australia Ltd Buy 1,723 1,78 3,75 1,78 3,75 1,78 3,75 1,78 3,72 1,78 3,75 3,6 3,6 1,78 3,78 3,6 3,6 3,7 3,6 3,6 3,6 3,7 3,6 3,6 3,6 3,7 3,6 3,6 3,7 3,6 3,6 3,7 3,6 3,6 3,7 3,6 3,6 3,7 3,6 3,6 3,7 3,6 3,6 3,7 3,6 3,6	3-Jul-14	UBS Securities Australia Ltd	Buy	978	101	Ordinary
UBS Securities Australia Ltd Buy 5.285 546 UBS Securities Australia Ltd Buy 5.285 546 UBS Securities Australia Ltd Buy 7,163 7.03 UBS Securities Australia Ltd Buy 17.259 1,783 UBS Securities Australia Ltd Buy 3,156 150 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 1,723 1,786 UBS Securities Australia Ltd Buy 2,724 287 UBS Securities Australia Ltd Buy 2,72 28 UBS Securities Australia Ltd Buy 2,73 25 UBS Securities Australia Ltd Buy 2,73 2,73 UBS Securities Australia Ltd Buy 2,73 2,74 UBS Securities Australia Ltd Buy 2,74 2,73 UBS Securities Australia Ltd Buy 2,74	3-Jul-14	UBS Securities Australia Ltd	Buy	4,172	431	Ordinary
UBS Securities Australia Ltd Buy 5,285 546 UBS Securities Australia Ltd Buy 7,163 740 UBS Securities Australia Ltd Buy 1,756 1,786 UBS Securities Australia Ltd Buy 3,156 1,256 UBS Securities Australia Ltd Buy 1,452 159 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 1,723 1,08 UBS Securities Australia Ltd Buy 5,74 6,97 UBS Securities Australia Ltd Buy 5,74 6,97 UBS Securities Australia Ltd Buy 2,73 1,08 UBS Securities Australia Ltd Buy 2,74 6,97 7,24 UBS Securities Australia Ltd Buy 2,041 2,774 1,08 UBS Securities Austra	3-Jul-14	UBS Securities Australia Ltd	Buy	5,285	246	Ordinary
UBS Securities Australia Ltd Buy 1,163 740 UBS Securities Australia Ltd Buy 1,7259 1,783 UBS Securities Australia Ltd Buy 3,156 366 UBS Securities Australia Ltd Buy 1,472 150 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 1,723 1,089 UBS Securities Australia Ltd Buy 2,52 26 UBS Securities Australia Ltd Buy 2,52 26 UBS Securities Australia Ltd Buy 2,53 2,74 UBS Securities Australia Ltd Buy 2,53 2,61 28 UBS Securities Australia Ltd Buy 2,53 2,61 28 UBS Securities Australia Ltd Buy 2,732 2,61 28 UBS Securi	3-Jul-14	UBS Securities Australia Ltd	Buv	5 285	546	Ordinary
UBS Securities Australia Ltd Buy 1,735 1,783 UBS Securities Australia Ltd Buy 1,725 1,783 UBS Securities Australia Ltd Buy 3,156 3,26 UBS Securities Australia Ltd Buy 5,024 5,19 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 5,74 945 UBS Securities Australia Ltd Buy 5,74 945 UBS Securities Australia Ltd Buy 5,74 6,97 UBS Securities Australia Ltd Buy 2,5,48 2,72 UBS Securities Australia Ltd Buy 2,041 2,77 UBS Securities Australia Ltd Buy 2,74 2,77 UBS Securities Australia Ltd Buy 2,74 2,72 UBS Securities Australia Ltd Buy	2.4.1.14	I IBC Contrition Australia I to	5.00	7.163	240	62
UBS Securities Australia Ltd Buy 1,7459 1,783 UBS Securities Australia Ltd Buy 474 49 UBS Securities Australia Ltd Buy 474 49 UBS Securities Australia Ltd Buy 2,778 159 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 1,723 1178 UBS Securities Australia Ltd Buy 1,723 1,083 UBS Securities Australia Ltd Buy 1,723 1,083 UBS Securities Australia Ltd Buy 2,52 26 UBS Securities Australia Ltd Buy 6,970 2,52 UBS Securities Australia Ltd Buy 6,970 7,20 UBS Securities Australia Ltd Buy 2,53 26 UBS Securities Australia Ltd Buy 2,732 2,69 UBS Securities Australia Ltd Buy 2,732 2,69 UBS Securities Australia Ltd Buy 3,146 3,76 UBS Securities Australia Ltd Buy 3,1	1 1 1	1 19C C++	, and	CD1./	7.10	Colliary
UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 4,452 169 UBS Securities Australia Ltd Buy 1,452 159 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 2,778 386 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 9,148 945 UBS Securities Australia Ltd Buy 9,148 945 UBS Securities Australia Ltd Buy 6,370 720 UBS Securities Australia Ltd Buy 26,390 727 UBS Securities Australia Ltd Buy 26,390 727 UBS Securities Australia Ltd Buy 26,390 27.724 UBS Securities Australia Ltd Buy 22,041 2.774 UBS Securities Australia Ltd Buy 3,146 376 UBS Securities Australia Ltd Buy 3,146 376 UBS Securities Australia Ltd Buy <	-101-14	UBS Securities Australia Ltd	Buy	657'/1	1,/83	Ordinary
UBS Securities Australia Ltd Buy 474 49 UBS Securities Australia Ltd Buy 5,024 150 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 1,181 122 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 1,723 1,08 UBS Securities Australia Ltd Buy 2,72 26 UBS Securities Australia Ltd Buy 2,73 26 UBS Securities Australia Ltd Buy 6,970 720 UBS Securities Australia Ltd Buy 2,53 26 UBS Securities Australia Ltd Buy 2,53 2,78 UBS Securities Australia Ltd Buy 2,53 2,78 UBS Securities Australia Ltd Buy 2,732 2,64 UBS Securities Australia Ltd Buy 2,732 2,64 UBS Securities Australia Ltd Buy 3,146 3,75 UBS Securities Australia Ltd Buy 3,146	3-Jul-14	UBS Securities Australia Ltd	Buy	3,156	326	Ordinary
UBS Securities Australia Ltd Buy 1,452 150 UBS Securities Australia Ltd Buy 2,778 257 UBS Securities Australia Ltd Buy 2,778 257 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 9,148 945 UBS Securities Australia Ltd Buy 6,747 56 UBS Securities Australia Ltd Buy 6,370 720 UBS Securities Australia Ltd Buy 6,370 720 UBS Securities Australia Ltd Buy 2,328 2,724 UBS Securities Australia Ltd Buy 2,328 2,724 UBS Securities Australia Ltd Buy 2,327 2,736 UBS Securities Australia Ltd Buy 2,732 2,736 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy <	8-Jul-14	UBS Securities Australia Ltd	Buy	474	49	Ordinary
UBS Securities Australia Ltd Buy 5,024 5,19 UBS Securities Australia Ltd Buy 1,78 287 UBS Securities Australia Ltd Buy 1,783 178 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 1,723 1,88 UBS Securities Australia Ltd Buy 5,42 1,689 UBS Securities Australia Ltd Buy 5,747 697 UBS Securities Australia Ltd Buy 26,368 2,724 UBS Securities Australia Ltd Buy 26,382 2,724 UBS Securities Australia Ltd Buy 26,382 2,724 UBS Securities Australia Ltd Buy 26,382 2,724 UBS Securities Australia Ltd Buy 2,041 2,277 UBS Securities Australia Ltd Buy 3,146 326 UBS Securities Australia Ltd Buy 3,146 326 UBS Securities Australia Ltd Buy 3,146 326 UBS Securities Australia Ltd Buy	8-101-14	UBS Securities Australia I td	2018	1 452	150	Ordinan
UBS Securities Australia Ltd Buy 2,724 517 UBS Securities Australia Ltd Buy 2,778 287 UBS Securities Australia Ltd Buy 1,181 122 UBS Securities Australia Ltd Buy 1,723 1,786 UBS Securities Australia Ltd Buy 1,622 1,089 UBS Securities Australia Ltd Buy 252 26 UBS Securities Australia Ltd Buy 25,306 27 UBS Securities Australia Ltd Buy 25,306 27 UBS Securities Australia Ltd Buy 23,307 27 UBS Securities Australia Ltd Buy 2,341 2,271 UBS Securities Australia Ltd Buy 2,341 2,277 UBS Securities Australia Ltd Buy 2,041 2,277 UBS Securities Australia Ltd Buy 3,146 326 UBS Securities Australia Ltd Buy 3,146 326 UBS Securities Australia Ltd Buy 2,759 265 UBS Securities Australia Ltd Buy <td< td=""><td>11 14</td><td>LIDE Commission Assets In 144</td><td>600</td><td>100</td><td>200</td><td>5 6</td></td<>	11 14	LIDE Commission Assets In 144	600	100	200	5 6
UBS Securities Australia Ltd Buy 1,73 287 UBS Securities Australia Ltd Buy 1,73 178 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 1,723 1,78 UBS Securities Australia Ltd Buy 9,148 945 UBS Securities Australia Ltd Buy 5,747 697 UBS Securities Australia Ltd Buy 2,53 2,6 UBS Securities Australia Ltd Buy 2,6,747 697 UBS Securities Australia Ltd Buy 2,6,747 697 UBS Securities Australia Ltd Buy 2,2,041 2,774 UBS Securities Australia Ltd Buy 2,0,041 2,777 UBS Securities Australia Ltd Buy 2,749 2,85 UBS Securities Australia Ltd Buy 2,749 2,85 UBS Securities Australia Ltd Buy 3,146 3,16 UBS Securities Australia Ltd Buy 3,156 3,16 UBS Securities Australia Ltd Buy	- Inc-	ODS Securities Australia Ltd	Buy	5,024	519	Ordinary
UBS Securities Australia Ltd Bby 1,181 122 UBS Securities Australia Ltd Bby 1,736 198 UBS Securities Australia Ltd Bby 1,733 178 UBS Securities Australia Ltd Bby 1,642 1,089 UBS Securities Australia Ltd Bby 2,52 26 UBS Securities Australia Ltd Bby 6,370 7,73 UBS Securities Australia Ltd Bby 6,370 7,20 UBS Securities Australia Ltd Bby 2,636 2,741 UBS Securities Australia Ltd Bby 2,631 2,88 UBS Securities Australia Ltd Bby 2,2041 2,277 UBS Securities Australia Ltd Bby 2,2041 2,277 UBS Securities Australia Ltd Bby 3,146 326 UBS Securities Australia Ltd Bby 3,146 326 UBS Securities Australia Ltd Bby 3,156 326 UBS Securities Australia Ltd Bby 3,146 326 UBS Securities Australia Ltd Bby	3-JUI- 14	UBS Securities Australia Ltd	Buy	2,178	287	Ordinary
UBS Securities Australia Ltd Bby 3,736 386 UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 9,148 945 UBS Securities Australia Ltd Buy 5,72 25 UBS Securities Australia Ltd Buy 6,370 720 UBS Securities Australia Ltd Buy 26,388 2,724 UBS Securities Australia Ltd Buy 26,370 720 UBS Securities Australia Ltd Buy 26,370 277 UBS Securities Australia Ltd Buy 22,041 2,77 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 3,156 285 UBS Securities Australia Ltd Buy 3,156 366 UBS Securities Australia Ltd Buy 3,156 366 UBS Securities Australia Ltd Buy 3,156 366 UBS Securities Australia Ltd Buy 2	3-Jul-14	UBS Securities Australia Ltd	Buy	1,181	122	Ordinary
UBS Securities Australia Ltd Buy 1,723 178 UBS Securities Australia Ltd Buy 10,542 1,089 UBS Securities Australia Ltd Buy 2,52 26 UBS Securities Australia Ltd Buy 6,747 697 UBS Securities Australia Ltd Buy 2,63,60 720 UBS Securities Australia Ltd Buy 2,63,60 720 UBS Securities Australia Ltd Buy 2,63,72 2,66 UBS Securities Australia Ltd Buy 2,041 2,277 UBS Securities Australia Ltd Buy 2,041 2,277 UBS Securities Australia Ltd Buy 3,146 3,26 UBS Securities Australia Ltd Buy 2,123 2,133 UBS Securities Australia Ltd Buy 2,132 2,133 UBS Securities Australia Ltd Buy <td>3-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>3,736</td> <td>386</td> <td>Ordinary</td>	3-Jul-14	UBS Securities Australia Ltd	Buy	3,736	386	Ordinary
UBS Securities Australia Ltd Bby 10,542 1,089 UBS Securities Australia Ltd Buy 9,148 945 UBS Securities Australia Ltd Buy 2,52 26 UBS Securities Australia Ltd Buy 6,77 57 UBS Securities Australia Ltd Buy 2,724 2,724 UBS Securities Australia Ltd Buy 2,670 2,724 UBS Securities Australia Ltd Buy 2,2041 2,277 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,759 2,193 UBS Securities Australia Ltd Buy 2,724 2,193 UBS Securities Australia Ltd Buy	8-Jul-14	UBS Securities Australia Itd	Birv	1 723	178	Ordinan
UBS Securities Australia Ltd Buy 1,027-6 UBS Securities Australia Ltd Buy 9,148 9,67 UBS Securities Australia Ltd Buy 26,368 274 697 UBS Securities Australia Ltd Buy 6,970 720 720 UBS Securities Australia Ltd Buy 26,388 2,724 2,004 2,277 UBS Securities Australia Ltd Buy 56,1 561 586 1,005 3,006 1,005	8. b.d. 14	I IBC Committee Australia I+d	,	573.01	000	6
UBS Securities Australia Ltd Bby 51,48 945 UBS Securities Australia Ltd Bby 252 26 UBS Securities Australia Ltd Bby 26,368 2,724 UBS Securities Australia Ltd Buy 6,747 697 UBS Securities Australia Ltd Buy 26,356 2,724 UBS Securities Australia Ltd Buy 22,322 2,306 UBS Securities Australia Ltd Buy 22,041 2,277 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,759 2,759 UBS Securities Australia Ltd Buy 2,759 2,759 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 2,728 2,193 UBS Securities Australia Ltd Buy		THE STORY OF THE PARTY OF THE P	, and	240,01	200,-	Clemary
UBS Securities Australia Ltd Bluy 552 26 UBS Securities Australia Ltd Buy 6,747 697 UBS Securities Australia Ltd Buy 26,368 2,724 UBS Securities Australia Ltd Buy 26,370 720 UBS Securities Australia Ltd Buy 22,322 2,306 UBS Securities Australia Ltd Buy 561 58 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 2,759 2,193 UBS Securities Australia Ltd Buy 2,759 2,193 UBS Securities Australia Ltd Buy	-1-In-C	ODS Seculines Australia Ltd	buy	9,148	345	Ordinary
UBS Securities Australia Ltd Buy 6,747 697 UBS Securities Australia Ltd Buy 2,536 2,724 UBS Securities Australia Ltd Buy 6,900 770 UBS Securities Australia Ltd Buy 5,601 770 UBS Securities Australia Ltd Buy 22,041 2,777 UBS Securities Australia Ltd Buy 22,041 2,277 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 8,131 840 UBS Securities Australia Ltd Buy 3,156 265 UBS Securities Australia Ltd Buy 2,759 266 UBS Securities Australia Ltd Buy 2,759 267 UBS Securities Australia Ltd Buy 2,758 2,133 UBS Securities Australia Ltd Buy	3-Jul-14	UBS Securities Australia Ltd	Buy	252	26	Ordinary
UBS Securities Australia Ltd Buy 26,368 2,724 UBS Securities Australia Ltd Buy 6,970 720 UBS Securities Australia Ltd Buy 20,332 2,366 UBS Securities Australia Ltd Buy 26,01 2,277 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,759 2,193 UBS Securities Australia Ltd Buy 2,1,228 2,193 UBS Securities Australia Ltd Buy 2,1,228 2,193 UBS Securities Australia Ltd Buy 2,1,228 2,193 UBS Securities Australia Ltd Buy 2,098 330 UBS Securities Australia Ltd Buy 2,098 330 UBS Securities Australia Ltd Buy 2,688 2,631 UBS Securities Australia Ltd Buy	3-Jul-14	UBS Securities Australia Ltd	Buy	6,747	269	Ordinary
UBS Securities Australia Ltd Buy 6,970 720 UBS Securities Australia Ltd Buy 2,5322 3.06 UBS Securities Australia Ltd Buy 561 58 UBS Securities Australia Ltd Buy 2,041 2,777 UBS Securities Australia Ltd Buy 3,146 356 UBS Securities Australia Ltd Buy 3,146 36 UBS Securities Australia Ltd Buy 8,131 840 UBS Securities Australia Ltd Buy 3,156 265 UBS Securities Australia Ltd Buy 2,759 286 UBS Securities Australia Ltd Buy 2,1,228 2,163 UBS Securities Australia Ltd Buy 2,1,228 2,163 UBS Securities Australia Ltd Buy 2,1,228 2,163 UBS Securities Australia Ltd Buy 2,759 2,163 UBS Securities Australia Ltd Buy 2,759 2,163 UBS Securities Australia Ltd Buy 2,758 2,163 UBS Securities Australia Ltd Buy	3-Jul-14	UBS Securities Australia Ltd	Bliv	26.368	2 724	Ordinary
UBS Securities Australia Ltd Buy 22,372 2,306 UBS Securities Australia Ltd Buy 22,322 2,306 UBS Securities Australia Ltd Buy 22,041 2,277 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 2,728 2,193 UBS Securities Australia Ltd Buy 2,728 2,193 UBS Securities Australia Ltd Buy 3,766 3,766 UBS Securities Australia Ltd Buy 2,728 2,193 UBS Securities Australia Ltd Buy 3,768 3,768 UBS Securities Australia Ltd Buy 2,748 2,631 UBS Securities Australia Ltd	11.17	1 (OC Committee August 1 to		050,03	7,7,2	O'Callial y
UBS Securities Australia Ltd BMy 2,522 306 UBS Securities Australia Ltd Buy 561 58 UBS Securities Australia Ltd Buy 2,041 2,277 UBS Securities Australia Ltd Buy 3,146 326 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 8,131 840 UBS Securities Australia Ltd Buy 3,156 265 UBS Securities Australia Ltd Buy 3,156 265 UBS Securities Australia Ltd Buy 2,1228 2,139 UBS Securities Australia Ltd Buy 2,1,228 2,193 UBS Securities Australia Ltd Buy 2,1,228 2,193 UBS Securities Australia Ltd Buy 2,1,228 2,193 UBS Securities Australia Ltd Buy 2,745 2,193 UBS Securities Australia Ltd Buy 2,748 2,193 UBS Securities Australia Ltd Buy 2,748 2,133 UBS Securities Australia Ltd Buy <td>-1-10-0</td> <td>Coo Securities Australia Lto</td> <td>ina</td> <td>0/6'9</td> <td>02/</td> <td>Ordinary</td>	-1-10-0	Coo Securities Australia Lto	ina	0/6'9	02/	Ordinary
UBS Securities Australia Ltd Buy 561 58 UBS Securities Australia Ltd Buy 22,041 2,277 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 8,131 840 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 2,728 2,193 UBS Securities Australia Ltd Buy 3,766 3,76 UBS Securities Australia Ltd Buy 2,728 2,193 UBS Securities Australia Ltd Buy 3,098 320 UBS Securities Australia Ltd Buy 2,468 2,631 UBS Securities Australia Ltd Buy 2,468 2,631	3-Jul-14	UBS Securities Australia Ltd	Buy	22,322	2,306	Ordinary
Buy 22,041 2,277 Buy 3,146 326 Buy 3,146 326 Buy 3,146 326 Buy 2,759 285 Buy 3,156 326 Buy 3,156 326 Buy 2,759 285 Buy 21,228 2,193 Buy 1,051 3096 Buy 25,468 2,631 Buy 25,468 2,631	3-Jul-14	UBS Securities Australia Ltd	Buy	561	28	Ordinary
UBS Securities Australia Ltd Buy 3,146 326 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 2,759 265 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 3,126 3,167 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 1,051 UBS Securities Australia Ltd Buy 3,098 320 UBS Securities Australia Ltd Buy 2,468 2,631 UBS Securities Australia Ltd Buy 2,468 2,631	8-11-14	I IRS Securities Australia I td	20.00	22.041	7776	Crain
UBS Securities Australia Ltd Buy 3.15b 3.2c UBS Securities Australia Ltd Buy 3.146 325 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 3.146 3.26 UBS Securities Australia Ltd Buy 3.156 3.76 UBS Securities Australia Ltd Buy 21,228 2,193 UBS Securities Australia Ltd Buy 10,174 1,051 UBS Securities Australia Ltd Buy 2,149 1,051 UBS Securities Australia Ltd Buy 2,648 3.546 UBS Securities Australia Ltd Buy 2,468 2,631	1	DI DI DI DI DI DI DI DI DI DI DI DI DI D	600	140,22	2,2,7	Cicilialy
UBS Securities Australia Ltd Buy 3,146 325 UBS Securities Australia Ltd Buy 8,131 840 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 3,156 326 UBS Securities Australia Ltd Buy 2,1228 2,193 UBS Securities Australia Ltd Buy 1,031 1,051 UBS Securities Australia Ltd Buy 2,5,468 2,631 UBS Securities Australia Ltd Buy 2,5,468 2,631	5-JUI-14	UBS Securities Australia Ltd	Buy	3,156	376	Ordinary
UBS Securities Australia Ltd Buy 8.131 840 UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 3,156 3,26 UBS Securities Australia Ltd Buy 2,1,228 2,193 UBS Securities Australia Ltd Buy 1,0,174 1,051 UBS Securities Australia Ltd Buy 2,4,68 3,20 UBS Securities Australia Ltd Buy 2,4,68 2,631	5-JUI-14	UBS Securities Australia Ltd	Buy	3,146	325	Ordinary
UBS Securities Australia Ltd Buy 2,759 285 UBS Securities Australia Ltd Buy 3,156 336 UBS Securities Australia Ltd Buy 2,728 2,193 UBS Securities Australia Ltd Buy 10,174 1,051 UBS Securities Australia Ltd Buy 2,631 320 UBS Securities Australia Ltd Buy 2,648 2,631	3-Jul-14	UBS Securities Australia Ltd	Buy	8,131	840	Ordinary
UBS Securities Australia Ltd Buy 3,156 376 UBS Securities Australia Ltd Buy 21,228 2,193 UBS Securities Australia Ltd Buy 10,174 1,051 UBS Securities Australia Ltd Buy 2,548 2,631 UBS Securities Australia Ltd Buy 2,548 2,631	3-Jul-14	UBS Securities Australia Ltd	Buv	2.759	285	Ordinary
USS Securities Australia Ltd Buy 2,128 2,139 UBS Securities Australia Ltd Buy 10,174 1,051 UBS Securities Australia Ltd Buy 10,174 1,051 UBS Securities Australia Ltd Buy 3,098 320 UBS Securities Australia Ltd Buy 25,468 2,631	3-1-11-14	I IRS Securities Australia I to	Sin	3 156	300	Ordina
Opp Securities Australia Ltd Buy 10,174 1,051 UBS Securities Australia Ltd Buy 10,174 1,051 UBS Securities Australia Ltd Buy 2,648 2,631 UBS Securities Australia Ltd Buy 2,648 2,631	11 14	De Constitut A settle	600	000.00	037	Cidillary
UBS Securities Australia Ltd Buy 10,174 1,051 UBS Securities Australia Ltd Buy 3,096 320 UBS Securities Australia Ltd Buy 25,468 2,631	#1-In-	Obs Securities Australia Ltd	lauy	877'17	2,193	Ordinary
UBS Securities Australia Ltd Buy 3,098 320 UBS Securities Australia Ltd Buy 25,468 2,631	3-Jul-14	UBS Securities Australia Ltd	Buy	10,174	1,051	Ordinary
UBS Securities Australia Ltd Buy 25,468 2,631	3-Jul-14	UBS Securities Australia Ltd	Buy	3,098	320	Ordinan
CDS SECULIARS AUSTRALIA LLU	2-11-11-14				247	200
	****	I IDC Congriting Arctivitia 140	7 d	25,750	25.5	Ordinaly

UBS Securities Australia Ltd	Buy	1,558	161	Ordinary
UBS Securities Australia Ltd	Buy	1,026	106	Ш
UBS Securities Australia Ltd	Buy B.c.	4,821	498	Ordinary
UBS Securities Australia Ltd	Buv	3311	347	Ordinary
UBS Securities Australia Ltd	Buy	288	9	Ordinary
ralia Ltd	Buy	8,809	910	
ralia Ltd	Buy	8,267	854	Ordinary
UBS Securities Australia Ltd	Buy	21,122	2,182	Ordinary
UBS Securities Australia Ltd	Buy Buy	18,944	1,957	Ordinary
LIBS Securities Australia Ltd	200	320	15.	Ordinary
tralia Ltd	Buv	2.585	267	Ordinary
UBS Securities Australia Ltd	Buy	165	17	Ordinary
UBS Securities Australia Ltd	Buy	4,830	499	Ordinary
UBS Securities Australia Ltd	Buy	781	29	Ordinary
UBS Securities Australia Ltd	Buy	2,217	529	Ordinary
UBS Securities Australia Ltd	Buy	958	66	Ordinary
ralia Ltd	Buy	2,517	260	Ordinary
ralia Ltd	Buy	34,277	3,541	Ordinary
UBS Securities Australia Ltd	Buy	2,585	792	Ordinary
UBS Securities Australia Ltd	Buy	4,840	200	Ordinary
UBS Securities Australia Ltd	Buy	1,355	140	Ordinary
Securities Australia Ltd	Buy	1,084	112	Ordinary
Securities Australia Ltd	Buy	3,010	311	Ordinary
UBS Securities Australia Ltd	Buy	2,633	272	Ordinary
UBS Securities Australia Ltd	Buy	2,585	267	Ordinary
UBS Securities Australia Ltd	Buy	3,359	347	Ordinary
UBS Securities Australia Ltd	Buy	2,575	266	Ordinary
Securities Australia Ltd	Buy	2,710	280	Ordinary
lia Ltd	Buy	689	99	Ordinary
UBS Securities Australia Ltd	Buy	1,433	148	Ordinary
UBS Securities Australia Ltd	Buy	145	15	Ordinary
UBS Securities Australia Ltd	Buy	826	66	Ordinary
UBS Securities Australia Ltd	Buy	2,846	294	Ordinary
UBS securities Australia Ltd	Buy	056'5	406	Ordinary
UBS securities Australia Ltd	Buy	23,464	2,424	Ordinary
ralia Ltd	Buy	090'6	936	Ordinary
UBS Securities Australia Ltd	Buy	067	30	Ordinary
Obsideration Australia Ltd	auy	27073	607	Ordinary
stralia Lto	Buy	8/8	SF S	Ordinary
UBS Securities Australia Ltd	Buy	18,566	1,918	Ordinary
UBS Securities Australia Ltd	Buy	855,95	2,820	Ordinary
stralia Ltd	Buy	4,869	203	Ordinary
UBS Securities Australia Ltd	Buy	4,704	486	Ordinary
UBS Securities Australia Ltd	Buy	2,207	228	Ordinary
UBS Securities Australia Ltd	Buy	2,284	236	Ordinary
Securities Australia Ltd	Buy	2,768	286	Ordinary
Securities Australia Ltd	Buy	2,739	283	Ordinary
JBS Securities Australia Ltd	Buy	2,633	272	Ordinary
JBS Securities Australia Ltd	Buy	2,633	272	Ordinary
JBS Securities Australia Ltd	Buv	3,543	396	Ordinary
JBS Securities Australia Ltd	Suv.	3.040	314	Ordinary
tralia I td	Biv	877.0	287	Ordinary
IRS Securities Australia I to	Bin	319	25	Ordinan
ממום דוס	, and	616	3	Cidilary
Securities Australia Ltd	ong.	soc'oc	1//'0	Crainary
UBS Securities Australia Ltd	Buy	1,888	195	Ordinary
UBS Securities Australia Ltd	Buy	13,668	1,412	Ordinary
Securities Australia Ltd	Buy	15,498	1,601	Ordinary
Securities Australia Ltd	Buy	8,218	849	Ordinary
UBS Securities Australia Ltd	Buv	54,934	5.675	Ordinary
stralia I td	> 20	8 131	840	Ordinary
UBS Securities Australia Ltd	N. G	4 162	430	Ordinary
V+1 **	ă	V00 C	000	Ordinan

08-101-14	100 Constitute Australia 14d			,	,
08-111-14	LIRS Securities Australia Ltd	D ouy	928	96	Ordinary
28-Jul-14	UBS Securities Australia Itd	Bird	15.469	1 502	Ordinary
18. hif. 14	I IRS Securities Australia 14d	, a	000 0	0601	Cidillary
10.0	Obs Securities Australia Ltd	Buy	4,840	2000	Ordinary
-10-00	UDS Securities Australia Ltd	Buy	13,358	1,381	Ordinary
- 1-in-o	ODS Securities Australia Ltd	Buy	4,240	438	Ordinary
08-JUI- 14	UBS Securities Australia Ltd	Buy	2,788	288	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	3,020	312	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	4,162	430	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	1,994	506	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buv	832	98	Ordinary
-111-14	LIBS Securities Australia Ltd	Bitk	2 707 8	200	i i i
11 11 20	1 IDC Countries Australia Ltd	ind.	10/15	200	Ordinary
100	Obs Securities Australia Liu	Ang	4,995	916	Ordinary
US-JUI-14	UBS Securities Australia Ltd	Buy	2,341	242	Ordinary
08-Jui-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	10	F	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buv	4.162	430	Ordinary
08-Jul-14	UBS Securities Australia I to	N. B	25.46	590	Croling
V -	I IBC Countition Australia 144	600	0.530	6	S S S S S S S S S S S S S S S S S S S
	COS Securities Australia Etc	, and	9,0,4	606	Ordinary
-Inr-oc	UBS Securities Australia Ltd	Buy	4,443	459	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	484	20	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Buy	1,394	144	Ordinary
38-111-14	LIBS Securities Australia Ltd	Bis	1 133	117	Croling
1.1.1	I IDC Constitute Australia tod	(5)		1	1
	COO Occultura Australia Lio	pu)	2 :	1	Ordinary
11-101-	ODS SECULINES AUSTRAINA LTD	Buy	01	-	Ordinary
28-3Ui-14	UBS Securities Australia Ltd	Buy	2,343	242	Ordinary
38-Jul-14	UBS Securities Australia Ltd	Buy	2,343	242	Ordinary
28-Jul-14	UBS Securities Australia Ltd	Buv	4.704	486	Ordinary
-11-14	I IRS Securities Australia I to	Bin	2 439	253	(in the contract of
1 1 1 1	LIDE Consisten Assessible 14-3	ína m	50472	767	Coursey
	Coo.	pag	4,000	404	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	1,355	140	Ordinary
08-Jul-14	UBS Securities Australia Ltd	Sell	2,202	(227)	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buv	35.011	3.678	Ordinary
11 11 00	Libo Cocimitation A section 1991	600	47.07	2007	(iii)
	Con Securities Australia Etc	136	400/14	(4,992)	Clainary
1	Obs securities Australia Liu	ació	441,015	40,439	Ordinary
09-JUI-14	UBS Securities Australia Ltd	Sell	238,670	(25,102)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	669,249	70,085	Ordinary
39-Jui-14	UBS Securities Australia Ltd	les.	26 960	(6,000)	Ordinary
09-bil-14	I IBS Societies Australia I to	100	808 C	(000)	
	COO Securities Australia City	126	2,030	(nnc)	Crainary
- Ini- Id	UBS Securities Australia Ltd	Buy	999'/56	101,090	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Sell	706,667	(74,243)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	5.202	548	Ordinary
-11-14	LIBS Securities Australia I to	Enter Into Derivative	ON A	150	, acital
10-li-l-14	I IRC Securities Australia 14d	Enter Into Delivative	0.52	2	Charles
	Obs securities Australia Ltd	CITIES INTO DELINGUAE	AW.	-	Options
J9-Jul-14	UBS Securities Australia Ltd	Enter Into Derivative (NA	150	Options
09-Jul-14	UBS Securities Australia Ltd	Enter Into Derivative C	AVA	150	Options
09-Jul-14	UBS Securities Australia 11d	Foter Into Derivative O	A/N	40	Ontions
4.1.1.4	Libe Manie Management Australia Ltd	i d	1000	200	2000
1	ODS Wealth Manageries & Abstralla Ltd	buy	1,431	507	Ordinary
09-Jul-14	UBS Wealth Management Australia Ltd	Buy	3,352	354	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	37,075	3,915	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buv	905.29	6.611	Ordinary
- hal-14	I IRS Securities Australia 14d	18	25 493	2,600	Ordingo
1 7	7 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	650	20075	70007	2000
-Jul-14	UBS SECURITIES AUSTRAIG LTO	BUÿ	U88'/E	4,000	Ordinary
J9-Jul-14	UBS Securities Australia Ltd	Buy	2,311	244	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	112,882	11.920	Ordinary
00-1-1-14	BS Courries Australia Ltd	D	710.250	75,000	Complete
1	COO SCOTLINGS AGAINST ELG	, and	002,017	000,5	Cidillary
J9-JUI-14	UBS Securities Australia Ltd	Buy	172,970	18,265	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	7,140	754	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buv	75.465	2.689	Ordinary
00 1.1 14	I IBC Congrition Anothrolia 144	P. S.	1 545	150	
	Dia September Caro	, and	010,1	201	Clustraly
Jul-14	UBS Securities Australia Ltd	Buy	511	54	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	375,552	39,657	Ordinary
09-Jul-14	UBS Securities Australia Ltd	l Buy	179,665	18.972	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buv	970 E	22.2	, dedical
	700 00000000000000000000000000000000000	,		//	A IPCIII A
	In 1997 the military Assessed in the	ľ	2012	326	Ordinary
09-JUI-14	UBS Securities Australia Ltd	Buy	205,139	21,662	Ordinary

<u> </u>	7 Secreties Australia Ita	NO.	12107	1.260	Ordinary
	Commission Assessed and	7-19	21 202		
08-Jul-14 0B:	UBS Securities Australia Ltd	buy B. II.	505,150	5,240	Ordinary
T	UBS Securities Australia Ind	B. W.	7 570	783	Ordinary
T	S Cognitive Australia I to	S and	565	70/	Ordinary
Ī	UBS Securities Australia I to	200	600	5 6	Ordinary
Т	UBS Securities Australia Ltd	Buv	368	38	Ordinary
T	Securities Australia I td	Buy	1 849	191	Ordinary
08-hil-14 UB	UBS Securities Australia Ltd	Buy	29,456	3.043	Ordinary
Γ	UBS Securities Australia Ltd	Buv	3.833	396	Ordinary
Ī	UBS Securities Australia Ltd	Buv	610	63	Ordinary
ı	S Securities Australia Ltd	Buy	24,897	2,572	Ordinary
1	S Securities Australia Ltd	Buy	43,521	4,496	Ordinary
08-Jul-14 UBS	UBS Securities Australia Ltd	Buy	12,371	1,278	Ordinary
	UBS Securities Australia Ltd	Buy	288	40	Ordinary
	5 Securities Australia Ltd	Buy	21,344	2,205	Ordinary
	UBS Securities Australia Ltd	Buy	12,845	1,327	Ordinary
08-Jul-14 UBS	S Securities Australia Ltd	Buy	21,064	2,176	Ordinary
- 1	UBS Securities Australia Ltd	Buy	1,539	159	Ordinary
	S Securities Australia Ltd	Buy	2,681	277	Ordinary
08-Jul-14 UBS	S Securities Australia Ltd	Buy	11,422	1,180	Ordinary
1	UBS Securities Australia Ltd	Buy	1,617	167	Ordinary
- 1	UBS Securities Australia Ltd	Buy	9,409	972	Ordinary
08-Jul-14 UBS	UBS Securities Australia Ltd	Buy	1,268	131	Ordinary
J	S Securities Australia Ltd	Buy	1,704	176	Ordinary
08-Jul-14 UBS	S Securities Australia Ltd	Buy	10,183	1,052	Ordinary
	UBS Securities Australia Ltd	Buy	4,840	200	Ordinary
	S Securities Australia Ltd	Buy	1,113	115	Ordinary
Ī	JBS Securities Australia Ltd	Buy	15,972	1,650	Ordinary
08-Jul-14 UBS	UBS Securities Australia Ltd	Buy	19,980	2,064	Ordinary
	S Securities Australia Ltd	Buy	37,268	3,850	Ordinary
	JBS Securities Australia Ltd	Buy	18,247	1,885	Ordinary
08-Jul-14 UBS	JBS Securities Australia Ltd	Buy	14,946	1,544	Ordinary
Γ	UBS Securities Australia Ltd	Buy	416	43	Ordinary
Ī	JBS Securities Australia Ltd	Buy	31,431	3,247	Ordinary
Ī	5 Securities Australia Ltd	Buy	6,263	647	Ordinary
Ī	UBS Securities Australia Ltd	Bov	6.447	999	Ordinary
Ī	IBS Securities Australia I td	Buy	4 337	448	Ordinary
T	IBS Societies Australia 11d	Bir	8 180	BAE	Ordinary
Ť	OCCUPATION AND AND AND AND AND AND AND AND AND AN	0.0	14 163	1 463	Ordinary
	UBS Securities Australia Lid	ond.	201,41	cor c	Ordinary
1	S Securities Australia Lid	and and	077'17	2,193	Ordinary
08-JUI- 14 UB	s securities Australia Ltd	gnà	4,927	806	Ordinary
1	UBS Securities Australia Ltd	gnà	1,404	145	Ordinary
-	s Securities Australia Ltd	Buy	12,081	1,248	Ordinary
	JBS Securities Australia Ltd	Buy	5,188	536	Ordinary
08-Jul-14 UBS	JBS Securities Australia Ltd	Buy	2,033	210	Ordinary
Ī	5 Securities Australia Ltd	Buy	7,570	782	Ordinary
Γ	UBS Securities Australia Ltd	Buv	26.833	2,772	Ordinary
Ī	IRS Securities Australia it d	Buy	7 734	489	Ordinary
Ī	2 - C. C. C. C. C. C. C. C. C. C. C. C. C.	650	000 /	203	Ordinan
T	Obsisted Hills Australia Edu	had	EC0'+	202	O'GHI (BI)
T	S Securities Australia Liu	, and	7,570	700	Ordinal y
T	UBS Securities Australia Ltd	any	/69	7/	Ordinary
08-Jul-14 UB	Securities Australia Ltd	suy	4,859	205	Ordinary
1	JBS Securities Australia Ltd	Buy	1,239	128	Ordinary
1	UBS Securities Australia Ltd	Buy	4,569	472	Ordinary
	5 Securities Australia Ltd	Buy	38,362	3,963	Ordinary
08-Jul-14 UBS	JBS Securities Australia Ltd	Buy	7,018	725	Ordinary
Ī	5 Securities Australia Ltd	Buy	1,897	196	Ordinary
	UBS Securities Australia Ltd	Buy	1,713	177	Ordinary
08-Jul-14 UBS	5 Securities Australia Ltd	Buy	20,986	2,168	Ordinary
08-Jul-14 UBS	UBS Securities Australia Ltd	Buy	1,636	169	Ordinary
	JBS Securities Australia Ltd	Buy	9,128	943	Ordinary
ĺ	JBS Securities Australia Ltd	Buy	1,713	177	Ordinary
Ī	JBS Securities Australia Ltd	Buy	17,211	1.778	Ordinary
T	IDC Commission Americal and	(55)		•	
		200	14 104	1 457	Ordinary

00-4-1-17	Late Character Assertation Co. 1				
09-101-14	I IRS Securities Australia (1)	Buy	135 130	07	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Bis	234.014	77 633	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	21,793	2 204	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	40.831	4 298	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	3.838	404	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	57	v	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	6 964	733	Ordinary
09-Jul-14	UBS Securities Australia Ltd	8110	83 686	008 &	ordina Variable
09-Jul-14	UBS Securities Australia Ltd	Birv	13,614	1 /33	, acuipo
9-Jul-14	UBS Securities Australia Ltd	NI B	3.800	00%	Ordinary.
09-11-14	I IBS Socierties Australia I to	200	1 843	100	Oldinary Ordinary
11.50	COS Securities Australia Lite	hng	7/9/1	<u>/6</u>	Ordinary
10.00	UDS Securities Australia Ltd	Buy	77/	9/	Ordinary
79-Jul-14	UBS Securities Australia I.td	Buy	28,472	2,997	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	5,311	529	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Виу	221,673	23,334	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	1,254	132	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	4,769	505	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buv	11,970	1.260	Ordinary
59-Jul-14	UBS Securities Australia Ltd	Buy	17,803	1,874	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	10,973	1.155	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buv	6.014	633	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	2 185	230	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buv	5 985	059	Ordinary
29-Jul-14	UBS Securities Australia Ltd	Buv	150.676	15.869	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Biv	010,000	- 100	Ordinary
29-Jul-14	UBS Securities Australia 11d	(S) 8	0.470	260	ordinary.
29-Jul-14	UBS Securities Australia 11d	Bir	532	222	Ordinary
9-1-1-14	LIRS Securities Australia (14	Sec. of	1 107	טר לי	Ciditaly
9-11-14	I BS Securities Australia itd	Buy	161,131	070 31	Ordinary Ordinary
79-1-1-14	I IBC Countries Australia 144	650	152,101	0,5,0	Couring
1 1 1	Libe committee Australia Ltd	śna	15,032	004,	Ordinary
20 1.1 14	Obs Securities Australia Liu	Buy	4,788	405	Ordinary
23-Jul-14	Obs Securities Australia Ltd	Buy	01/,1	180	Ordinary
41-12-0	UBS Securities Australia Ltd	Buy	3,800	400	Ordinary
79-Jul-14	UBS Securities Australia Ltd	Buy	922	65	Ordinary
9-Jui-14	UBS Securities Australia Ltd	Buy	105	11	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	874	35	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	25	9	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	2,565	270	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	11,438	1.204	Ordinary
29-Jul-14	UBS Securities Australia Ltd	Buv	77.064	8 112	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	95	101	Ordinary
09-1111-14	I BS Securities Australia I to	Bury	4 375	2 0	Ordinary Crains
09-1-11-14	LIRS Securities Australia 11d	Birv	0,1	2	Grainary
11/1/1	LIDS Countries Australia Ltd	Son S	2 1	7	Orginary
03-101-14	UBS Securities Australia Ltd	Buy	515'/	/91	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Вчу	6,802	716	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	5,187	546	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	17,005	1,790	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	1,264	133	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	5,520	581	Ordinary
29-Jul-14	UBS Securities Australia Ltd	Buv	29	6	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Ris	19779	2 082	Ordinary
3-11-14	I IRS Securities Australia Ltd	Buch	533	200/4	Ordings
00-1-1-14	I IBS Contrition Australia Ltd	600	12 213	200	Oldinary
09-1-1-14	I IRS Securities Australia 14d	Son a	2120	25.0	Ordinary
5-11-14-24	I IRS Securities Australia 14d	, and	101 745	012.01	Ordinary
11-11-00	I IRC Countries Australia 1+4	, and	0300	01/10	Ordinary
1.1 14	Line C mitter Australia Ltd	Buy	2,330	CSC.	Ordinary
79-Jul-14	UBS Securities Australia Ltd	Buy	48	2	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	95,333	10,035	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	979	103	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	3,800	400	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	5,558	585	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	£	O.	Cadinan
				2	Cramary
09-Jul-14	UBS Securities Australia Ltd	Su.	1.653	174	Ordinary

Obsideration Australia Ltd	, g	OF.C	Ç	
UBS Securities Australia Ltd	Riiv	715 37	2 779	Ordinary
JBS Securities Australia Ltd	Buv	89 965	9.500	Ordinary
equities Australia I to	8117	80,004	00000	Ordinan
IDS Consider Australia to	626	00,000	000	O Commission
Contries Australia Ltd		626,00	060,0	Oldinary
ODS Securities Australia Ltd	Boy	cos'os	Sec's	Ordinary
curities Australia Ltd	hay	74,380	2,635	Ordinary
scurities Australia Ltd	Buy	23,463	2,475	Ordinary
curities Australia Ltd	Buy	1,782	188	Ordinary
UBS Securities Australia Ltd	Buy	32,004	3,376	Ordinary
scurities Australia Ltd	Buv	1.583	167	Ordinary
IBS Securities Australia 14d	Bisk	5 070	630	Ordinan
The last of the last of the last	65	1001		(Displayed)
Obs securities Australia Liu	had	180,1	4	Ordinary
ecurities Australia Ltd	Buy	7,868	830	Ordinary
UBS Securities Australia Ltd	Buy	16,230	1,712	Ordinary
ecurities Australia I Id	Si.v	0269	730	Ordinary
A Contract Acceptable 144	7.0	017 67	COLV	
ecurities Australia Lid	pný	43,418	4,580	Orginary
ecurities Australia Ltd	Buy	790'9	640	Ordinary
UBS Securities Australia Ltd	Buy	29,284	3,089	Ordinary
ecurities Australia Ltd	Buy	4,029	425	Ordinary
JBS Securities Australia Ltd	Buv	32.118	3 388	Ordinary
A traiting Americal 144	, and	0	C	Ordinan
מפס ספרמו וופס אתפון פוופ דות	na)	6	7	Clumary
ecurities Australia Ltd	Buy	3,849	406	Ordinary
ecurities Australia Ltd	Buy	1,488	157	Ordinary
I IBS Securities Australia I to	S _I S	57 771	8008	Ordinan
bt leiter Australia 14d	8.0	14.761	1 556	Ording
	600	1000	2001	0.00
Obs securities Australia Liu	, and	26/16	400	Ordinary
JBS Securities Australia Ltd	Buy	7,232	0,51	Ordinary
UBS Securities Australia Ltd	Buy	8,361	882	Ordinary
UBS Securities Australia Ltd	BCV	290'9	640	Ordinary
Politifies Australia # td	Bin	5 972	630	Ordinary
Att distance A society	0.30	020.00	2242	Orași e e
P. D. Constitute Australia 140	7.0	0.000 6	070	, in the second
econities wasti alia Eta	, and	COC'C	0/0	Clumary
JBS Securities Australia Ltd	, and	5		Ordinary
UBS Securities Australia Ltd	Buy	4,42/	46/	Ordinary
UBS Securities Australia Ltd	Buy	3,591	378	Ordinary
ecurities Australia Ltd	Buy	4,722	497	Ordinary
ecurities Australia Ltd	Buy	53	3	Ordinary
on urities Δustralia i td	Biv	2 242	950	Ordinary
and the second second	6.6		107	Control of
UDS Securities Australia Ltd	, inc.	205,1	13/	Ordinary
ecurities Australia Ltd	Suy	14,250	1,500	Ordinary
ecurities Australia Ltd	Buy	1,045	110	Ordinary
ecurities Australia Ltd	Buy	8,541	668	Ordinary
equitips Australia I td	Buy	3.021	318	Ordinary
eculines Australia Liv	, and	120,0	010	S C
ecurifies Australia Lid	pay	/61/1	97	Ordinary
UBS Securities Australia Ltd	Buy	2,185	230	Ordinary
ecurities Australia Ltd	Buy	32,728	3,445	Ordinary
ecurities Australia Ltd	Buv	2.955	311	Ordinary
I BS Sequirities Australia I td	8114	44 840	4 720	Ordinary
Att cleans a ceitic con Odl	8,11	030 77	000	Overline C
Accurates Australia Ltd	áng.	007/11	000,1	Oldinary Coldinary
UBS Securities Australia Ltd	guy	13,120	1,381	Ordinary
UBS Securities Australia Ltd	Buy	2,850	8	Ordinary
JBS Securities Australia Ltd	Buy	11,400	1.200	Ordinary
IRC Contrition Australia 14d	Birv	11 250	1 500	Codingo
Contract Acceptance		010.4.4	005.	O'ding)
ODS SECULIES AUSTRAIG EIG	ína c	002/41	000,1	5 6
UBS Securities Australia Ltd	Buy	188,6	619	Ordinary
UBS Securities Australia Ltd	Buy	11,400	1,200	Ordinary
UBS Securities Australia Ltd	Buy	3,886	409	Ordinary
Politritips Australia I td	Ruy	2 983	314	Ordinary
ODD December Address and Education	600	2000	1001	O CHILD
scurities Australia Ltd	Buy	/77'69	/97'/	Ordinary
JBS Securities Australia Ltd	Buy	72,390	7,620	Ordinary
UBS Securities Australia Ltd	Buy	13,243	1,394	Ordinary
UBS Securities Australia Ltd	Buv	24.016	2.528	Ordinary
At the second section of	Britis	11 501	1 210	(Carried)
Obs securities Australia Ltd	had	100,1	5.7.	in the second second
on mittee Δuetralia (td			1.1.	Ordinary
	- Buy	190	20	Ordinary

1111	Opp Securities Australia Lita		2,18/	240	Crainary
111	ILIBS Sequirities Australia I to	Riv	136.3	029	ording.
	UBS Securities Australia Ltd	Buy	5.928	624	Ordinary
Н	UBS Securities Australia Ltd	Buy	361	38	Ordinary
I	UBS Securities Australia Ltd	Buv	10	-	Ordinary
	UBS Securities Australia Ltd	Buy	57	9	Ordinary
	UBS Securities Australia Ltd	Buy	38	4	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	686'8	883	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	12,350	1,300	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	2,138	225	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	7,572	767	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	31,464	3,312	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	694	73	Ordinary
29-Jul-14	UBS Securities Australia Ltd	Buy	22,420	2,360	Ordinary
١	UBS Securities Australia Ltd	Buy	51,215	5,391	Ordinary
1	UBS Securities Australia Ltd	Buy	1,262	133	Ordinary
1	UBS Securities Australia Ltd	Buy	28,622	3,016	Ordinary
	UBS Securities Australia Ltd	Buy	28,138	2,965	Ordinary
09-301-14	UBS Securities Australia Ltd	Buy	5,390	268	Ordinary
11-10-00	Upo Courting Australia Ltd	buy Pr	28,265	3,010	Ordinary
100-00	UBS Securities Australia Ltd	buy Buss	121,5	332	Ordinary
-111-12	I IRS Sequities Australia 14d	Busy	30,000	3,240	Ordinary
-Jul-14	UBS Securities Australia Itd	Buy	7177	2,300	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	25.015	2 636	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buv	σ	7,72	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buv	1.215	128	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	228	24	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	14,244	1,501	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	40,351	4,252	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	36,945	3,893	Ordinary
19-Jul-14	UBS Securities Australia Ltd	Buy	10,923	1,151	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	23,393	2,465	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	27,720	2,921	Ordinary
09-Jul-14	UBS Securities Australia t.td	Buy	29,912	3,152	Ordinary
-101-14	UBS Securities Australia Ltd	Buy	23,725	2,500	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	2,856	301	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	2,068	534	Ordinary
- 1	UBS Securities Australia Ltd	Buy	24,892	2,623	Ordinary
- 1	UBS Securities Australia Ltd	8uy	2,942	310	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	27,749	2,924	Ordinary
- 1	UBS Securities Australia Ltd	Buy	668'9	727	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	4,327	456	Ordinary
09-Jui-14	UBS Securities Australia Ltd	Buy	2,846	300	Ordinary
	UBS Securities Australia Ltd	Buy	1,651	174	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	719,731	75,841	Ordinary
	UBS Securities Australia Ltd	Buy	3,796	400	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	5,182	546	Ordinary
	UBS Securities Australia Ltd	Buy	2,809	596	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	3,796	400	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	6,254	659	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	10,382	1,094	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	4,859	512	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	30,634	3,228	Ordinary
Γ	UBS Securities Australia Ltd	Buv	1,509	150	Ordinary
Τ	UBS Securities Australia Ltd	Buy	550 b	954	Ordinary
T	UBS Securities Australia Ltd	Buv	29.561	3 115	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buv	3 659	386	Ordinary
Τ	UBS Securities Australia Ltd	Bery	32 598	3 435	Ordinary
19-Jul-14	UBS Securities Australia Ltd	Buv	78.027	8,222	Ordinary
Γ	UBS Securities Australia Ltd	Buy	29,704	3,130	Ordinary
	UBS Securities Australia Ltd	Buy	5,979	630	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	27,720	2,921	Ordinary
	UBS Securities Australia Ltd	Buy	14,226	1,499	Ordinary
Г	UBS Securities Australia Ltd	Buv	26.648	2 RNR	Ordinary
	ODD Deculines Australia Ltd	(na	242/04	2,808	Ordinary

Buy 26,876 26,876 Buy 133 14 Buy 133 14 Buy 12,833 12 Buy 1,216 20 Buy 1,216 20 Buy 1,216 3 Buy 1,250 1, Buy 1,20 1, Buy 1,20 1, Buy 1,20 1, Buy 1,50 1, Buy 1,20 1, Buy 1,20 1, Buy 1,20 1, Buy 1,20 1, Buy 1,230 1, Buy 1,00	26,876 26,876 1,139 11,137 11,1370 11,	829 Ordinary		14,294 Ordinary	14 Ordinary			935 Ordinary	L	500 Ordinary	L			527 Ordinary		5 Ordinary			704 Ordinary	1	235 Ordinary			9 Ordinary		L	Ц			82 Ordinary		Ĺ		400 Ordinary		500 Ordinary					(56) Ordinary		Ĺ.						403 Ordinary			L			_		Ordinary
Buy Buy	Buy Buy	L			133		٤		2						l	135								80																										5							3,002 7,885 3,002 316
		26	6	135				100	25	4	7	11	9	14			126	164	16	32	21	22	4		15		12	3,	2,	28	105		Ř	e,	3	4				5,	7	,	20,	22,	3,	•	9,	-	, ,	56,	46,			10,		/4,	74, 3,
		2	- A	yr.	y.	λr	λr :	4		4	ŕ	y.	Á	Ar	2. 2	2		λí	Á	∆ r ≥		λí	χ.	Á.		2	ıy.	A	A	×	. >	. A	γ.	۸	Α.	. >								_					- >			, .	y	٨			
ies Australia Ltd ies Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd	18	i i ii	Β̈́	ă	ă ,	8 6	8	S E	- B	Bu	BI	<u>8</u>	8	20 0	20 00	8	Bu	8	ñα	3 8	Bu	- B	B	38	Bu	Bu	- B	ng d	9 8	200	Bu	Bu	- B	98	78	Se	Se	S	Se	as as	Sel.	Sel	Sel	S.	Se	as S	20	- E	B	Bu	Bu	Bu	Bu	å	31.	B Buy
	UBS Securiti UBS S	ies Australia Ltd	ies Australia Ltd	ies Australia Ltd	ies Australia Ltd	ies Australia Ltd	les Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	ies Australia Ltd	ies Australia Ltd	es Australia Ltd	ies Australia Ltd	les Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia I to	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	ac Arictrafia i to	A Adamana Live	UBS Securities Australia Ltd IRS Securities Australia Ltd

1111	Opp permittee Australia Ltd				
	TO CONTRACTOR ALICENSIS AND	ouy B	97/50	711	Ordinary
\mathbf{I}	UBS Securities Australia Ind	Buy	30,149	3,18/	Ordinary
ŀ	TIDE Committee Australia Ltd	buy D	917	57	Orginary
	UBS Securities Australia Ltd	Buy	2,157	228	Ordinary
1	UBS Securities Australia Ltd	Buy	4,371	462	Ordinary
09-Jul-14	UBS Securties Australia Ltd	Buy	2,800	296	Ordinary
- 1	UBS Securities Australia Ltd	Buy	3,576	378	Ordinary
- 1	UBS Securities Australia Ltd	Buy	3,784	400	Ordinary
Į	UBS Securities Australia Ltd	Buy	096'5	930	Ordinary
	UBS Securities Australia Ltd	Buy	32,505	3,436	Ordinary
	UBS Securities Australia Ltd	Buv	16 167	1 700	Ordinary
ı	UBS Securities Australia I td	Riv	000	100	in in
09-Jul-14	UBS Securities Australia Ltd	Rim	195,53	000.3	O Gillian
T	I BC Coourities Australia 1-d	, in a	290,00	00,40	Clumary
T	UDC Consister Australia Ltd	buy D	3,784	400	Ordinary
T	UBS Securities Australia Ltd	Buy	2,960	630	Ordinary
1	UBS Securities Australia Ltd	Buy	57,753	6,105	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	18,901	1,998	Ordinary
	UBS Securities Australia Ltd	Buy	688	94	Ordinary
٦	UBS Securities Australia Ltd	Buy	5,629	265	Ordinary
	UBS Securities Australia Ltd	Buy	1,220	129	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	10,851	1.147	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Riv	3 841	301	Ordina.
T	LIBS Securities Australia I to	Blox	000	200	Order lary
T	LIBS Societies Australia I to	Bin	000	000	Courtain
T	I IDC Commission Australia 14d	ouy Pro:	678'c	262	Orginary
Ť	OBS Securities Australia LIG	Buy	679'5	265	Ordinary
Ť	obs securities Australia Ltd	Buy	99	^	Ordinary
٦	UBS Securities Australia Ltd	Buy	4,475	473	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	14,956	1,581	Ordinary
_	UBS Securities Australia Ltd	Buy	4,484	474	Ordinary
Ť	UBS Securities Australia Ltd	Buv	45.389	4.798	Ordinary
Ī	UBS Securities Australia Itd	N. P.	1 551	164	Ordinan
Ī	I IDC Cocurition Australia 1+4		1001	5 8	S C C C C
Ť	on paralles Australia tio	had	cus'ı	138	Ordinary
1	Obsise Australia Ltd	Buy	15,789	1,669	Ordinary
	UBS Securities Australia Ltd	Buy	3,784	400	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	8,949	946	Ordinary
	UBS Securities Australia Ltd	Buy	757	80	Ordinary
_	UBS Securities Australia Ltd	Buy	341	36	Ordinary
09-Jui-14	UBS Securities Australia Ltd	Buv	4 106	434	Ordinan
T	JBS Securities Australia Ltd	Buy	10000	100	S C
Ť	IDC Completion A retrails 144		#80'01	000,1	Coloraly
Ť	Obsider Australia Ltd	Buy	7,261	239	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	4,484	474	Ordinary
	UBS Securities Australia Ltd	Buy	4,106	434	Ordinary
Г	UBS Securities Australia Ltd	Buv	2,129	225	Ordinary
Γ	UBS Securities Australia I to	Raix	200 282	21 172) are all poor
T	IDC Cocumition Americal 14d	9,	/87/007	7/1/17	Callary
Ť	Obsider Australia Liu	puy	4,475	4/3	Ordinary
09-Jul-14	JBS Securties Australia Ltd	Buy	896'8	948	Ordinary
	JBS Securities Australia Ltd	Buy	3,784	400	Ordinary
	JBS Securities Australia Ltd	Buy	5,742	209	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buv	1,656	175	Ordinary
Ī	UBS Securities Australia Ltd	Rity	1366	230	Ordinan
14-14-14	IRC Securities Australia I to	D. I.	27.254	1000	(1)
T	UBC Cocurition Australia Hol	b.g.	457'/7	188,7	Ordinary
#1-int-60	DDS Securities Australia Liu	buy	4,541	480	Ordinary
1	JBS Securities Australia Ltd	Buy	152,571	16,128	Ordinary
	UBS Securities Australia Ltd	Buy	60,345	6,379	Ordinary
_	JBS Securities Australia Ltd	Buy	62:328	606'9	Ordinary
Ī	UBS Securities Australia Ltd	Rev	2 185	231	Ordinary
T	JBS Sequities Australia I to	Rin	52.12		Ordinary
11 14	DO Commission A control of	900	/6	0	Cidinary
T	DDS Securities Australia Ltd	any.	4,4/5	4/3	Ordinary
Ť	Job Securities Australia Lto	Buy	49,968	2,782	Ordinary
1	JBS Securities Australia Ltd	Buy	435	46	Ordinary
	UBS Securities Australia Ltd	Buy	4,484	474	Ordinary
	JBS Securities Australia Ltd	Buy	4,531	479	Ordinary
09-Jul-14 C	UBS Securities Australia Ltd	Buy	7,388	781	Ordinary
T	UBS Securities Australia Itd	Bir	87 917	0 203) de dipo
T	no continue propaga and	ina i	715/10	9,293	Ordinary
7	UBS Securities Australia Ltd	Buy	586'69	7,398	Ordinary
09-Jul-14	IRS Securities Australia Ltd	Rin	2 157	228	Ordinary

Г	1				_				_	т-	T	_	1	Т	_	_	_	т	1		7	_	_	_	ī	-	1	_						_	_	- 1 -	_	_	_	_	_	т-	_		T	_	_	r-		_			_				_	_			_	_	_		-	_
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinany	Ordinary	Guinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Crainary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	CELLETY
400	740	125	629	624	80	1,730	300	833	2	4	-	89	2 000	245	2640	2,049	/3	8/8	01	879	96/	490	400	300	-	624	150	2	439	624	1,232	2	71	188	13	16/	'n	27 147	43/47/	f g	3 289	62	630	-	2,500	9,404	60,941	-	8,989	5,924	146	270	146	299'6	8,003	85	2,181	523	334	2	9,141	13	8,161	7	0 8	250
3,796	791,4	1,186	6,254	5,922	76	16,418	2,847	7,905	19	38	104	644	19 878	2007	980 30	7000	I SOL	3,580	5	5,947	7,538	4,640	3,788	2,841	6	5,909	1,421	47	4,157	5,909	11,667	61	672	8,400	123	1,491	350	27.5	707	101	31.147	748	2,966	6	23,638	88,962	576,502	o	92,036	56,041	1,381	2,554	1,381	91,450	75,708	804	20,632	4,948	3,160	61	86,474	123	77,203	2 2	560 0	2,636
Buy	an)	Buy	38	Buv	S S	600	200	and o	an)	Ang C	and Buy	áng.	Buy	ng a	and a	P. IV	200 a	Bisv	Buv	Buy	Buy	Buy	Buy	Buy	Buy	8uy	Buy	A G	Ruy	griy	Buy	Buy	and																																	
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IRS Societies Australia Ind	I BS Securities Australia Ltd	LIBC Con within Australia 14d	Obsider Australia Lid	Obsiderations Abstraina Ltd	USS Securities Australia Ltd	Obsideration of the control of the c	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Lid	IDS Contribor Australia Ltd	IRS Societies Australia 114	IRS Securities Australia 11d	IBS Securities Australia I to	J8S Securities Australia Itd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	DS Securities Australia Ltd	UBS Securities Australia Ltd	ארוים שומיזים שחזונים שווים ארוון ווים ארווי
09-Jul-14	00-501-14	03-3ul-14	99-101-14	09-Jul-14	09-101-14	09-Jul-14				09-Jul-14	ı	09-Jul-14	09-11-14	09-1-1-14	09-1-1-14	11 14 00	00-50I-14	00-301-14	1	1	09-101-14	1	1	09-Jul-14	-1	-	09-Jul-14	1	09-Jul-14	T	Τ	09-101-14	09-1-1-14	09-111-14	09-Jul-14	09-Jul-14	1	09-Jul-14		l			09-Jul-14	Ш	09-Jul-14	09-Jul-14		09-Jul-14	٦		Ī	1	09-jul-14	1	1	ı	1	09-Jul-14								

UBS Searchies Australia Ltd Buy 1,618 171 UBS Searchies Australia Ltd Buy 1,618 171 UBS Searchies Australia Ltd Buy 1,621 1,714 UBS Searchies Australia Ltd Buy 1,621 1,774 UBS Searchies Australia Ltd Buy 1,622 1,622 UBS Searchies Australia Ltd Buy 1,622 1,522 UBS Searchies Australia Ltd Buy 1,622 1,724 UBS Searchies Australia Ltd Buy 1,622 1,524 UBS Searchies Australia Ltd Buy 1,622 1,524 UBS Searchies Australia Ltd Buy 1,622 1,724 UBS Searchies Australia Ltd Buy 1,628 3,01 1,227 UBS Searchies Australia Ltd Buy 2,270 1,271	09-Jul-14	UBS Securities Australia Ltd	Buy	6		Ordinary
UBS Securities Australia Ltd Buy 6.15 6.65 UBS Securities Australia Ltd Buy 6.61 6.65 UBS Securities Australia Ltd Buy 1.07.14 1.77.14 UBS Securities Australia Ltd Buy 1.62.24 1.77.14 UBS Securities Australia Ltd Buy 1.62.73 1.62.73 UBS Securities Australia Ltd Buy 1.62.73 1.77.14 UBS Securities Australia Ltd Buy 1.62.73 1.77.14 UBS Securities Australia Ltd Buy 1.62.75 1.62.75	09-Jul-14	UBS Securities Australia Ltd	Buy	1,618	171	Ordinary
UBS Securities Australia Ltd Buy 16,214 1774 UBS Securities Australia Ltd Buy 16,214 1774 UBS Securities Australia Ltd Buy 16,214 1779 UBS Securities Australia Ltd Buy 16,214 1779 UBS Securities Australia Ltd Buy 2,626 1552 UBS Securities Australia Ltd Buy 3,001 3,001 3,001 UBS Securities Australia Ltd Buy 3,001 3,001 3,001 17,001 UBS Securities Australia Ltd Buy 3,001 3,001 3,001 17,001	09-Jul-14	UBS Securities Australia Ltd	Buy	615	99	Ordinary
UBS Securities Australia Ltd Buy 16,214 17791 UBS Securities Australia Ltd Buy 1,625 1,666 UBS Securities Australia Ltd Buy 1,682 1,666 UBS Securities Australia Ltd Buy 1,682 1,666 UBS Securities Australia Ltd Buy 2,889 30.93 UBS Securities Australia Ltd Buy 1,787 1,784 UBS Securities Australia Ltd Buy 1,787 1,784 UBS Securities Australia Ltd Buy 1,787 1,784 UBS Securities Australia Ltd Buy 1,784 1,784 UBS Securities Australia Ltd Buy 1,784 1,784 UBS Securities Australia Ltd Buy 2,296 2,297 UBS Securities Australia Ltd Buy 2,296 2,296 UBS Securities Australia Ltd	09-Jul-14	UBS Securities Australia Ltd	Buy	8,382	988	Ordinary
UBS Securities Australia Lud Buy 17,019 17,929 UBS Securities Australia Lud Buy 16,062 15,521 UBS Securities Australia Lud Buy 3,000 3,000 UBS Securities Australia Lud Buy 1,060 1,074 UBS Securities Australia Lud Buy 1,080 1,071 UBS Securities Australia Lud Buy 1,080 1,071 UBS Securities Australia Lud Buy 1,080 1,080 UBS Securities Australia Lud	09-Jul-14	UBS Securities Australia Ltd	Buy	16,214	1,714	Ordinary
UBS Securites Australia Lud Bay 1628 UBS Securites Australia Lud Boy 1,680 UBS Securites Australia Lud Boy 2,688 300 UBS Securities Australia Lud Boy 2,688 300 UBS Securities Australia Lud Boy 3,584 3,90 UBS Securities Australia Lud Boy 3,584 3,90 UBS Securities Australia Lud Boy 3,50 8,25 UBS Securities Australia Lud Boy 3,50 1,584 UBS Securities Australia Lud Boy 3,50 3,50 UBS Securities Australia Lud Boy 3,50 3,50 UBS Securities Australia Lud Boy 3,50 3,50 UBS Securities Australia Lud Boy 3,50 3,00 UBS Securities Australia Lud Boy 3,50 3,00 UBS Securities Australia Lud Boy 3,00 3,00 UBS Securities Australia Lud Boy 3,00 3,00 UBS Securities Australia Lud Boy 3,00 3,00 <tr< td=""><td>09-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>17.019</td><td>1.799</td><td>Ordinary</td></tr<>	09-Jul-14	UBS Securities Australia Ltd	Buy	17.019	1.799	Ordinary
UBS Securities Australia Ltd Bay 14,882 1,552 UBS Securities Australia Ltd Buy 3,604 381 UBS Securities Australia Ltd Buy 2,688 393 UBS Securities Australia Ltd Buy 3,305 3,490 UBS Securities Australia Ltd Buy 3,305 3,490 UBS Securities Australia Ltd Buy 3,305 3,490 UBS Securities Australia Ltd Buy 7,007 3,207 UBS Securities Australia Ltd Buy 7,007 3,207 UBS Securities Australia Ltd Buy 1,007 1,106 UBS Securities Australia Ltd Buy 1,007 1,107 UBS Securities Australia Ltd Buy 2,003 3,207 UBS Securities Australia Ltd Buy 2,003 3,00 UBS Securities Australia Ltd Buy<	09-Jul-14	UBS Securities Australia Ltd	BCv	624	99	Ordinary
USS Securities Australia Ltd Bay 3,604 381 USS Securities Australia Ltd Buy 2,838 300 USS Securities Australia Ltd Buy 9,344 30,490 USS Securities Australia Ltd Buy 3,490 1,667 USS Securities Australia Ltd Buy 3,490 1,667 USS Securities Australia Ltd Buy 1,067 1,244 USS Securities Australia Ltd Buy 1,067 1,244 USS Securities Australia Ltd Buy 1,067 1,244 USS Securities Australia Ltd Buy 1,069 1,277 USS Securities Australia Ltd Buy 1,060 1,277 USS Securities Australia Ltd Buy 1,060 1,277 USS Securities Australia Ltd Buy 1,060 1,077 USS Securities Australia Ltd Buy 1,060 1,077 USS Securities Australia Ltd Buy 2,206 2,206 USS Securities Australia Ltd Buy 2,206 3,415 USS Securities Australia Ltd	09-Jul-14	UBS Securities Australia Ltd	Buv	14 682	1 552	Ordinary
UBS Securities Autorials Ltd Bly	09-Jul-14	UBS Securities Australia Ltd	Buv	3 604	381	Ordinary
UBS Securities Australia Ltd Buy 2,828 300 UBS Securities Australia Ltd Buy 3,315 3,490 UBS Securities Australia Ltd Buy 1,987 1,865 UBS Securities Australia Ltd Buy 1,987 1,864 UBS Securities Australia Ltd Buy 1,087 1,264 UBS Securities Australia Ltd Buy 1,087 1,264 UBS Securities Australia Ltd Buy 1,087 1,264 UBS Securities Australia Ltd Buy 2,091 2,291 UBS Securities Australia Ltd Buy 2,256 2,21 UBS Securities Australia Ltd Buy 2,256 2,21 UBS Securities Australia Ltd Buy 2,256 2,21 UBS Securities Australia Ltd Buy 2,256 2,26 UBS Securities Australia Ltd Buy 2,256 2,40 UBS Securities Australia Ltd Buy 2,526 2,52 UBS Securities Australia Ltd Buy 2,526 2,52 UBS Securities Australia Ltd Buy <td>09-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>0</td> <td>-</td> <td>Ordinary</td>	09-Jul-14	UBS Securities Australia Ltd	Buv	0	-	Ordinary
UBS Securities Australia Ltd Buy 3,394 993 UBS Securities Australia Ltd Buy 3,1190 3,397 1,16 UBS Securities Australia Ltd Buy 1,197 1,189 1,197 1,187 1,187 1,184 1,197 1,187 1,186 1,197 1,187 1,186 1,197 1,187 1,186 1,197 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187 1,186 1,187	09-Jul-14	UBS Securities Australia Ltd	Buy	7 838	OU.	Ordina
UBS Securities Australia Ltd Buy 3,315 3,490 UBS Securities Australia Ltd Buy 1,697 1,867 UBS Securities Australia Ltd Buy 1,997 1,264 UBS Securities Australia Ltd Buy 1,090 1,277 UBS Securities Australia Ltd Buy 1,090 1,277 UBS Securities Australia Ltd Buy 2,091 2,091 UBS Securities Australia Ltd Buy 2,091 3,091 UBS Securities Australia Ltd Buy 2,091 3,091 UBS Securities Australia Ltd Buy 2,092 3,091 UBS Securities Australia Ltd Buy 2,092 3,092 UBS Securities Australia Ltd <	09-4-14	BS Counties Australia 14d	D. 10	7000	000	1
UBS Securities Australia Ltd Buy 7,805 116 UBS Securities Australia Ltd Buy 7,805 116 UBS Securities Australia Ltd Buy 1,097 1,1597 116 UBS Securities Australia Ltd Buy 1,097 1,164 1,172 UBS Securities Australia Ltd Buy 1,646 1,174 UBS Securities Australia Ltd Buy 2,201 2,201 UBS Securities Australia Ltd Buy 2,206 2,201 UBS Securities Australia Ltd Buy 2,206 2,207 UBS Securities Australia Ltd Buy 2,206 3,415 UBS Securities Australia Ltd Buy 2,506 3,415 UBS Securities Australia Ltd Buy 2,506 3,415 UBS Securities Australia Ltd Buy 2,506 3,415 UB	00-1-1-1	I IDA Con without Avertalia I to	ouy D	9,094		Organary
UBS Securities Australia Ltd Buy 1,897 1,854 UBS Securities Australia Ltd Buy 1,957 1,564 UBS Securities Australia Ltd Buy 1,1997 1,277 UBS Securities Australia Ltd Buy 1,266 1,771 UBS Securities Australia Ltd Buy 2,093 624 UBS Securities Australia Ltd Buy 2,003 624 UBS Securities Australia Ltd Buy 2,003 624 UBS Securities Australia Ltd Buy 2,003 1,508 1,70 UBS Securities Australia Ltd Buy 2,003 624 624 UBS Securities Australia Ltd Buy 2,003 1,508 1,508 1,509 1,100 1,100 1,100 1,100 1,100 1,100 1,100 1,100 1,100 1,100 1,100 <	11 11	DO Commiss Australia Liu	ono.	53,015		Ordinary
UBS Securities Australia Ltd Bluy 11,097 1,1564 UBS Securities Australia Ltd Bluy 11,097 1,1564 UBS Securities Australia Ltd Bluy 1,204 1,237 UBS Securities Australia Ltd Bluy 2,093 6,24 UBS Securities Australia Ltd Bluy 2,266 2,67 UBS Securities Australia Ltd Bluy 3,068 3,415 UBS Securities Australia Ltd Bluy 3,068 3,425 UBS Securities Australia Ltd<	+1-Inc-60	ODS Securities Australia Ltd	BUY	7,805		Ordinary
UBS Securities Australia Ltd Bluy 11,957 1,564 UBS Securities Australia Ltd Bluy 11,957 1,264 UBS Securities Australia Ltd Bluy 1,666 1,273 UBS Securities Australia Ltd Bluy 2,091 2,271 UBS Securities Australia Ltd Bluy 2,091 2,271 UBS Securities Australia Ltd Bluy 2,071 2,27 UBS Securities Australia Ltd Bluy 2,27 2,27 UBS Securities Australia Ltd Bluy 2,27 2,27 UBS Securities Australia Ltd Bluy 2,503 6,23 UBS Securities Australia Ltd Bluy 2,503 2,50 UBS Securities Australia Ltd Bluy 2,503 2,50 UBS Securities Australia Ltd Bluy 2,503 6,24 UBS Securities Australia Ltd Bluy 3,06 3,115 UBS Securities Australia Ltd Bluy 3,06 3,20 UBS Securities Australia Ltd Bluy 3,06 3,20 UBS Securities Australia Ltd	09-JUI-14	UBS Securities Australia Ltd	Buy	1,097		Ordinary
UBS Securities Australia Ltd Buy 13,190 1,277 UBS Securities Australia Ltd Buy 1,646 1,721 UBS Securities Australia Ltd Buy 2,691 6,241 UBS Securities Australia Ltd Buy 5,993 6,241 UBS Securities Australia Ltd Buy 5,993 6,241 UBS Securities Australia Ltd Buy 5,270 2,401 UBS Securities Australia Ltd Buy 2,270 2,621 UBS Securities Australia Ltd Buy 2,270 2,621 UBS Securities Australia Ltd Buy 2,526 2,61 UBS Securities Australia Ltd Buy 2,503 6,524 UBS Securities Australia Ltd Buy 2,503 6,524 UBS Securities Australia Ltd Buy 3,656 3,415 UBS Securities Australia Ltd Buy 3,656 3,451 UBS Securities Australia Ltd Buy 5,603 6,528 UBS Securities Australia Ltd Buy 5,603 6,528 UBS Securities Australia Ltd <	09-Jul-14	UBS Securities Australia Ltd	Buy	11,957		Ordinary
UBS Securities Australia Ltd Buy 1,200 1,277 UBS Securities Australia Ltd Buy 1,266 1,271 UBS Securities Australia Ltd Buy 2,091 624 UBS Securities Australia Ltd Buy 5,903 624 UBS Securities Australia Ltd Buy 5,903 624 UBS Securities Australia Ltd Buy 2,270 2,40 UBS Securities Australia Ltd Buy 2,270 2,40 UBS Securities Australia Ltd Buy 2,20 2,40 UBS Securities Australia Ltd Buy 2,20 2,40 UBS Securities Australia Ltd Buy 3,60 3,415 UBS Securities Australia Ltd Buy	09-Jul-14	UBS Securities Australia Ltd	Buy	31,190		Ordinary
UBS Securities Australia Ltd Buy 1,646 1144 UBS Securities Australia Ltd Buy 2,091 2,21 UBS Securities Australia Ltd Buy 5,003 624 UBS Securities Australia Ltd Buy 5,003 624 UBS Securities Australia Ltd Buy 2,270 240 UBS Securities Australia Ltd Buy 2,270 241 UBS Securities Australia Ltd Buy 2,503 3,24 UBS Securities Australia Ltd Buy 3,056 3,03 UBS Securities Australia Ltd Buy 3,056 3,03 UBS Securities Australia Ltd Buy 3,056 3,03 UBS Securities Australia Ltd Buy 3,056 3,05 UBS Securities Australia Ltd Buy 3,056 3,05 UBS Securities Australia Ltd Buy	09-Jui-14	UBS Securities Australia Ltd	Buy	12,080		Ordinary
UBS Securities Australia Ltd Bby 2,091 221 UBS Securities Australia Ltd Buy 5,093 624 UBS Securities Australia Ltd Buy 5,093 624 UBS Securities Australia Ltd Buy 5,003 624 UBS Securities Australia Ltd Buy 2,270 2,40 UBS Securities Australia Ltd Buy 2,27 2,41 UBS Securities Australia Ltd Buy 2,25 2,52 UBS Securities Australia Ltd Buy 2,25 2,52 UBS Securities Australia Ltd Buy 3,63 3,23 UBS Securities Australia Ltd Buy 3,68 3,415 UBS Securities Australia Ltd Buy 3,68 3,68 3,68 UBS Securities Australia Ltd Buy 3,68 3,68 3,68 3,68 UBS Securities Australia Ltd Buy 3,68 3,68 3,68 3,68 UBS Securities Australia Ltd Buy 3,68 3,68 3,68 3,68 3,68 UBS Securities Australia	09-Jul-14	UBS Securities Australia Ltd	Buv	1 646		Ordinary
UBS Securities Australia Ltd Buy 5.902 6.24 UBS Securities Australia Ltd Buy 6.91 7.3 UBS Securities Australia Ltd Buy 2.270 2.40 UBS Securities Australia Ltd Buy 2.270 2.41 UBS Securities Australia Ltd Buy 2.270 2.62 UBS Securities Australia Ltd Buy 2.276 2.62 UBS Securities Australia Ltd Buy 2.266 2.27 UBS Securities Australia Ltd Buy 2.266 3.415 UBS Securities Australia Ltd Buy 3.066 3.415 UBS Securities Australia Ltd Buy 3.066 3.415 UBS Securities Australia Ltd Buy 3.066 3.70 UBS Securities Australia Ltd Buy 3.066 3.70 UBS Securities Australia Ltd Buy 5.066 3.70 UBS Securities Australia Ltd Buy 5.066 3.20 UBS Securities Australia Ltd Buy 5.066 3.20 UBS Securities Australia Ltd Buy	09-Jul-14	LIBS Securities Australia Ltd	χ. Σ.	100 6		Coling
UBS Securities Australia Ltd Buy 691 73 UBS Securities Australia Ltd Buy 691 73 UBS Securities Australia Ltd Buy 227 24 UBS Securities Australia Ltd Buy 227 24 UBS Securities Australia Ltd Buy 227 24 UBS Securities Australia Ltd Buy 75,256 267 UBS Securities Australia Ltd Buy 40,68 430 32 UBS Securities Australia Ltd Buy 4,08 303 32 UBS Securities Australia Ltd Buy 4,08 4,08 430 UBS Securities Australia Ltd Buy 5,09 3,09 3,09 UBS Securities Australia Ltd Buy 5,106 3,09 4,01 UBS Securities Australia Ltd Buy 5,106 3,09 4,01 UBS Securities Australia Ltd Buy 8,09 3,06 4,528 UBS Securities Australia Ltd Buy 8,09 3,06 4,20 UBS Securities Australia Ltd <t< td=""><td>09-1-1-14</td><td>I BS Securities Australia 14d</td><td>0.00</td><td>5003</td><td></td><td>Colliany</td></t<>	09-1-1-14	I BS Securities Australia 14d	0.00	5003		Colliany
UBS Securities Australia Ltd Buy 398 41 UBS Securities Australia Ltd Buy 389 43 UBS Securities Australia Ltd Buy 2270 240 UBS Securities Australia Ltd Buy 145,041 15,332 UBS Securities Australia Ltd Buy 30,3 32,7 24 UBS Securities Australia Ltd Buy 30,3 32,7 25 UBS Securities Australia Ltd Buy 30,3 32,7 25 UBS Securities Australia Ltd Buy 30,0 30,0 32,0 62,4 UBS Securities Australia Ltd Buy 30,0 30,0 32,0 62,4 UBS Securities Australia Ltd Buy 4,0 30,0 30,0 30,0 30,0 UBS Securities Australia Ltd Buy 4,0 30,0 30,0 30,0 40,0 UBS Securities Australia Ltd Buy 4,0 30,0 4,0 30,0 40,0 UBS Securities Australia Ltd Buy 6,5 30,0 30,0	00-1-1-14	LIDE Commission Australia ted	200	505,0		o deliai
Los Securities Australia Ltd Buy 1,2,20 24 UBS Securities Australia Ltd Buy 1,5,041 15,332 UBS Securities Australia Ltd Buy 7,6 8 UBS Securities Australia Ltd Buy 7,6 8 UBS Securities Australia Ltd Buy 2,256 2,57 UBS Securities Australia Ltd Buy 2,56 2,57 UBS Securities Australia Ltd Buy 2,68 3,93 UBS Securities Australia Ltd Buy 3,689 3,90 UBS Securities Australia Ltd Buy 3,689 4,528 UBS Securities Australia Ltd Buy 4,088 4,528 UBS Securities Australia Ltd Buy <t< td=""><td>1 1 00</td><td>The Countries Australia to</td><td>, and</td><td>160</td><td>6/</td><td>Ordinary</td></t<>	1 1 00	The Countries Australia to	, and	160	6/	Ordinary
UBS Securities Australia Ltd BMy 1,270 2,40 UBS Securities Australia Ltd BMy 1,270 24 UBS Securities Australia Ltd BMy 75 8 UBS Securities Australia Ltd BMy 2,276 267 UBS Securities Australia Ltd BMy 2,276 267 UBS Securities Australia Ltd BMy 2,276 267 UBS Securities Australia Ltd BMy 2,276 3,276 UBS Securities Australia Ltd BMy 3,689 390 UBS Securities Australia Ltd BMy 4,068 430 UBS Securities Australia Ltd BMy 4,068 430 UBS Securities Australia Ltd BMy 4,068 3,01 UBS Securities Australia Ltd BMy 4,068 3,01 UBS Securities Australia Ltd BMy 4,068 3,01 UBS Securities Australia Ltd BMy 4,072 4,07 UBS Securities Australia Ltd BMy 4,072 4,07 UBS Securities Australia Ltd BMy 7,16	09-Jul-14	Obs securities Australia Ltd	gnà	388	41	Ordinary
UBS Securities Australia Ltd Buy 145,041 15,332 UBS Securities Australia Ltd Buy 76 8 UBS Securities Australia Ltd Buy 76 8 UBS Securities Australia Ltd Buy 2,256 257 UBS Securities Australia Ltd Buy 2,568 303 322 UBS Securities Australia Ltd Buy 5,903 654 UBS Securities Australia Ltd Buy 5,903 654 UBS Securities Australia Ltd Buy 3,608 3,008 UBS Securities Australia Ltd Buy 3,608 3,008 UBS Securities Australia Ltd Buy 3,608 3,008 UBS Securities Australia Ltd Buy 8,508 3,008 UBS Securities Australia Ltd Buy 8,508 3,008 UBS Securities Australia Ltd Buy 8,508 3,009 UBS Securities Australia Ltd Buy 8,508 3,009 UBS Securities Australia Ltd Buy 8,208 3,009 UBS Securities Australia Ltd	09-Jul-14	UBS Securities Australia Ltd	Buy	2,270	240	Ordinary
UBS Securities Australia Ltd Buy 227 24 UBS Securities Australia Ltd Buy 2.506 26 UBS Securities Australia Ltd Buy 2.506 26 UBS Securities Australia Ltd Buy 2.508 2.508 UBS Securities Australia Ltd Buy 3.689 3.90 UBS Securities Australia Ltd Buy 3.689 3.23 UBS Securities Australia Ltd Buy 3.286 3.23 UBS Securities Australia Ltd Buy 4.068 3.74 UBS Securities Australia Ltd Buy 5.05 3.74 UBS Securities Australia Ltd Buy 5.05 3.75 UBS Securities Australia Ltd Buy 5.05 3.76 3.76 UBS Securities Australia Ltd Buy 5.05 3.76 3.78 UBS Securities Australia	09-Jul-14	UBS Securities Australia Ltd	Buy	145,041	15,332	Ordinary
UBS Securities Australia Ltd Buy 76 B UBS Securities Australia Ltd Buy 2,526 267 UBS Securities Australia Ltd Buy 2,526 267 UBS Securities Australia Ltd Buy 2,526 267 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 3,686 330 UBS Securities Australia Ltd Buy 4,068 340 UBS Securities Australia Ltd Buy 4,068 390 UBS Securities Australia Ltd Buy 4,068 3,056 UBS Securities Australia Ltd Buy 6,575 6,657 UBS Securities Australia Ltd Buy 7,840 4,702 UBS Securities Australia Ltd Buy <td< td=""><td>09-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>722</td><td>24</td><td>Ordinary</td></td<>	09-Jul-14	UBS Securities Australia Ltd	Buy	722	24	Ordinary
UBS Securities Australia Ltd Buy 2,526 267 UBS Securities Australia Ltd Buy 2,526 267 UBS Securities Australia Ltd Buy 3,689 32 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 3,689 303 UBS Securities Australia Ltd Buy 3,689 303 UBS Securities Australia Ltd Buy 3,689 303 UBS Securities Australia Ltd Buy 3,656 3,415 UBS Securities Australia Ltd Buy 3,656 3,415 UBS Securities Australia Ltd Buy 6,575 6,575 UBS Securities Australia Ltd Buy 6,575 6,575 UBS Securities Australia Ltd Buy 6,575 6,575 UBS Securities Australia Ltd Buy 6,575 6,58 UBS Securities Australia Ltd Buy 6,575 6,58 UBS Securities Australia Ltd Buy 4,788 1,580 UBS Securities Australia Ltd Buy	09-Jul-14	UBS Securities Australia Ltd	Buy	9/	80	Ordinary
UBS Securities Australia Ltd Buy 2,526 267 UBS Securities Australia Ltd Buy 47 5 UBS Securities Australia Ltd Buy 5,903 624 UBS Securities Australia Ltd Buy 5,903 624 UBS Securities Australia Ltd Buy 32,306 3,415 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 4,688 390 UBS Securities Australia Ltd Buy 6,575 695 UBS Securities Australia Ltd Buy 1,670 4,300 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 7,841 1,260 UBS Securities Australia Ltd Buy 3,687 1,260 UBS Securities Australia Ltd Buy 3	09-Jul-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
UBS Securities Australia Ltd Buy 47 5 UBS Securities Australia Ltd Buy 5,903 524 UBS Securities Australia Ltd Buy 5,903 624 UBS Securities Australia Ltd Buy 1,608 170 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 3,689 3,93 UBS Securities Australia Ltd Buy 4,068 430 UBS Securities Australia Ltd Buy 4,068 430 UBS Securities Australia Ltd Buy 6,575 695 UBS Securities Australia Ltd Buy 6,575 546 UBS Securities Australia Ltd Buy 7,650 3,734 UBS Securities Australia Ltd Buy 6,772 4,30 UBS Securities Australia Ltd Buy 7,015 </td <td>09-Jul-14</td> <td>UBS Securities Australia I td</td> <td>Bis</td> <td>3556</td> <td>7,967</td> <td>Ordinan</td>	09-Jul-14	UBS Securities Australia I td	Bis	3556	7,967	Ordinan
USS Securities Australia Ltd Buy 303 32 USS Securities Australia Ltd Buy 5,903 624 UBS Securities Australia Ltd Buy 1,608 1,00 UBS Securities Australia Ltd Buy 32,306 3,415 UBS Securities Australia Ltd Buy 4,068 3,05 UBS Securities Australia Ltd Buy 6,575 9 UBS Securities Australia Ltd Buy 6,575 9 UBS Securities Australia Ltd Buy 6,575 9 UBS Securities Australia Ltd Buy 6,575 5,66 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 3,78 3,78 UBS Securities Australia Ltd Buy 3,69 3,78 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072	09-1-1-14	LIBS Sequities Australia 11d	B _{i IV}	27.77		in diameter
UBS Securities Australia Ltd Buy 5,903 624 UBS Securities Australia Ltd Buy 5,903 624 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 4,068 3,415 UBS Securities Australia Ltd Buy 6,575 6,695 UBS Securities Australia Ltd Buy 6,576 5,46 UBS Securities Australia Ltd Buy 7,165 5,46 UBS Securities Australia Ltd Buy 7,841 4,072 4,30 UBS Securities Australia Ltd Buy 7,692 3,718 4,072 4,072 4,07 UBS Securities Australia Ltd Buy 7,010 3,718	09-1-14	I IRS Securities Australia 14d	6.00	4	0 00	Coloring
UBS Securities Australia Ltd Buy 5,903 100 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 3,056 3,236 UBS Securities Australia Ltd Buy 4,068 430 UBS Securities Australia Ltd Buy 6,575 655 UBS Securities Australia Ltd Buy 6,575 655 UBS Securities Australia Ltd Buy 6,575 655 UBS Securities Australia Ltd Buy 6,575 656 UBS Securities Australia Ltd Buy 6,575 5,66 5,46 UBS Securities Australia Ltd Buy 6,728 3,897 6,65 5,46 5,46 6,575 6,65 5,46 6,67 1,531 1,1533 9,46 6,575 6,67 1,531 1,66 3,415 1,1533 1,66 3,415 6,67 1,430 1,66 3,415 1,66 3,415 1,66 3,415 <td< td=""><td>1 1 00</td><td>Upo Cominco Australia Cio</td><td>, and</td><td>coc .</td><td>32</td><td>Ordinary</td></td<>	1 1 00	Upo Cominco Australia Cio	, and	coc .	32	Ordinary
UBS Securities Australia Ltd Buy 1,608 170 UBS Securities Australia Ltd Buy 3,689 390 UBS Securities Australia Ltd Buy 4,668 430 UBS Securities Australia Ltd Buy 4,668 430 UBS Securities Australia Ltd Buy 6,575 655 3,415 UBS Securities Australia Ltd Buy 6,575 695 UBS Securities Australia Ltd Buy 6,575 6,657 UBS Securities Australia Ltd Buy 6,575 6,65 UBS Securities Australia Ltd Buy 6,576 5,46 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 7,8,412 8,28 UBS Securities Australia Ltd Buy 7,8,412 8,28 UBS Securities Australia Ltd Buy 7,8,412 8,28 UBS Securities Australia Ltd Buy 7,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd	03-Jul-14	Obs securities Australia Lid	Ana	5,903	P24	Ordinary
UBS Securities Australia Ltd Bby 3,689 390 UBS Securities Australia Ltd Buy 3,689 340 UBS Securities Australia Ltd Buy 4,088 430 UBS Securities Australia Ltd Buy 4,088 430 UBS Securities Australia Ltd Buy 6,575 659 1 UBS Securities Australia Ltd Buy 6,575 669 546 UBS Securities Australia Ltd Buy 6,575 1,631 669 669 UBS Securities Australia Ltd Buy 8,765 5,46 546 669 UBS Securities Australia Ltd Buy 7,628 869 4,528 669 UBS Securities Australia Ltd Buy 7,641 8,260 3,734 7,73 4,90 1,533 1,560 1,533 1,560 1,533 1,534 1,528 1,560 1,534 1,560 1,534 1,534 1,528 1,560 1,534 1,534 1,534 1,534 1,528 1,560 1,534 1,560	09-Jul-14	UBS securities Australia Ltd	Buy	1,608	170	Ordinary
UBS Securities Australia Ltd Buy 4,068 3,415 UBS Securities Australia Ltd Buy 4,068 3,33 UBS Securities Australia Ltd Buy 4,068 3,33 UBS Securities Australia Ltd Buy 6,575 6,59 UBS Securities Australia Ltd Buy 6,575 6,69 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 3,290 4,280 UBS Securities Australia Ltd Buy 3,290 3,78 UBS Securities Australia Ltd Buy 7,641 1,260 UBS Securities Australia Ltd Buy 7,641 1,260 UBS Securities Australia Ltd Buy 7,671 4,072 UBS Securities Australia Ltd Buy 7,010 3,718 UBS Securities Australia Ltd Buy 7,010 3,718 UBS Securities Australia Ltd Buy </td <td>09-jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>3,689</td> <td>330</td> <td>Ordinary</td>	09-jul-14	UBS Securities Australia Ltd	Buy	3,689	330	Ordinary
UBS Securities Australia Ltd Buy 4,068 430 UBS Securities Australia Ltd Buy 3,056 3,23 UBS Securities Australia Ltd Buy 6,575 695 UBS Securities Australia Ltd Buy 6,577 695 UBS Securities Australia Ltd Buy 6,577 695 UBS Securities Australia Ltd Buy 6,577 695 UBS Securities Australia Ltd Buy 5,165 546 UBS Securities Australia Ltd Buy 3,587 3897 UBS Securities Australia Ltd Buy 3,585 3,897 UBS Securities Australia Ltd Buy 1,538 3,897 UBS Securities Australia Ltd Buy 1,092,18 1,1,533 UBS Securities Australia Ltd Buy 1,092,18 1,1,533 UBS Securities Australia Ltd Buy 6,072 4,072 UBS Securities Australia Ltd Buy 6,072 4,072 UBS Securities Australia Ltd Buy 6,072 2,016 UBS Securities Australia Ltd B	09-Jul-14	UBS Securities Australia Ltd	Buy	32,306	3,415	Ordinary
UBS Securities Australia Ltd Buy 3,056 323 UBS Securities Australia Ltd Buy 47 5 UBS Securities Australia Ltd Buy 6,575 65 UBS Securities Australia Ltd Buy 6,575 65 UBS Securities Australia Ltd Buy 6,575 65 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 42,880 4,528 UBS Securities Australia Ltd Buy 8,229 869 UBS Securities Australia Ltd Buy 42,880 4,528 UBS Securities Australia Ltd Buy 10,218 1,153 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 20,105 1,28 UBS Securities Australia Ltd Buy 20,105 1,28 UBS Securities Australia Ltd Buy <	09-Jul-14	UBS Securities Australia Ltd	Buy	4,068	430	Ordinary
UBS Securities Australia Ltd Buy 47 5 UBS Securities Australia Ltd Buy 6575 699 UBS Securities Australia Ltd Buy 6575 699 UBS Securities Australia Ltd Buy 1,657 546 UBS Securities Australia Ltd Buy 5,165 546 UBS Securities Australia Ltd Buy 5,165 546 UBS Securities Australia Ltd Buy 4,289 4,580 UBS Securities Australia Ltd Buy 36,905 3,897 UBS Securities Australia Ltd Buy 36,209 4,300 UBS Securities Australia Ltd Buy 1,592 1,260 UBS Securities Australia Ltd Buy 7,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 2,18 9,18 UBS Securities Australia Ltd Buy 2,010 2,123 UBS Securities Australia Ltd Buy 2,010 3,718 UBS Securities Australia Ltd Buy <td< td=""><td>09-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buv</td><td>3.056</td><td>323</td><td>Ordinary</td></td<>	09-Jul-14	UBS Securities Australia Ltd	Buv	3.056	323	Ordinary
UBS Securities Australia Ltd Buy 6578 9 UBS Securities Australia Ltd Buy 6,577 6,695 UBS Securities Australia Ltd Buy 6,577 6,695 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 3,580 4,528 UBS Securities Australia Ltd Buy 42,880 4,528 UBS Securities Australia Ltd Buy 8,229 889 UBS Securities Australia Ltd Buy 1,631 1,600 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 66,406 70,687 UBS Securities Australia Ltd Buy 66,406 70,73 UBS Securities Australia Ltd Buy 66,406 70,687 UBS Securities Australia Ltd Buy 66,406 70,687 UBS Securities Australia Ltd	09-Jul-14	UBS Securities Australia Ltd	Buv	47	4	Ordinary
UBS Securities Australia Ltd Buy 65 75 9 1 UBS Securities Australia Ltd Buy 6575 665 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 3,595 3,897 UBS Securities Australia Ltd Buy 3,590 3,897 UBS Securities Australia Ltd Buy 4,780 4,780 UBS Securities Australia Ltd Buy 7,841 1,280 UBS Securities Australia Ltd Buy 7,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 20,105 3,718 UBS Securities Australia Ltd Buy 20,105 3,718 UBS Securities Australia Ltd Buy </td <td>09-11-14</td> <td>I IRS Secretifies Australia Ltd</td> <td>Buy</td> <td>0</td> <td>,</td> <td>or in the</td>	09-11-14	I IRS Secretifies Australia Ltd	Buy	0	,	or in the
UBS Securities Australia Ltd Buy 6,575 695 UBS Securities Australia Ltd Buy 6,575 695 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 5,165 5,46 UBS Securities Australia Ltd Buy 5,165 5,48 UBS Securities Australia Ltd Buy 3,580 3,59 UBS Securities Australia Ltd Buy 78,412 11,533 UBS Securities Australia Ltd Buy 78,412 8,280 UBS Securities Australia Ltd Buy 78,412 8,280 UBS Securities Australia Ltd Buy 70,72 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 20,105 2,133 UBS Securities Australia Ltd Buy 20,105 2,134 UBS Securities Australia Ltd Buy 26,571 2,848 UBS Securities Australia Ltd B	00 11 14	L+1 cilculation A society and I	100	200	- 0	S S S S S S S S S S S S S S S S S S S
UBS Securities Australia Ltd Buy 6575 685 UBS Securities Australia Ltd Buy 18.267 1.918 UBS Securities Australia Ltd Buy 5,165 5.46 UBS Securities Australia Ltd Buy 5,165 5.48 UBS Securities Australia Ltd Buy 3,897 3,897 UBS Securities Australia Ltd Buy 8,229 899 UBS Securities Australia Ltd Buy 10,92,18 11,533 UBS Securities Australia Ltd Buy 10,92,18 11,533 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 5,105 2,148 UBS Securities Australia Ltd Buy 65,040 7,0687 UBS Securities Australia Ltd Buy 20,105 1,23 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd	11.00	Obsider Australia tio	ouy.	82	5	Ordinary
UBS Securities Australia Ltd Buy 19.31 UBS Securities Australia Ltd Buy 5.165 5.46 UBS Securities Australia Ltd Buy 5.165 5.46 UBS Securities Australia Ltd Buy 4.2880 4.288 UBS Securities Australia Ltd Buy 4.2880 4.288 UBS Securities Australia Ltd Buy 10.32 1.260 UBS Securities Australia Ltd Buy 78,412 8.280 UBS Securities Australia Ltd Buy 78,412 8.280 UBS Securities Australia Ltd Buy 7.6,412 8.280 UBS Securities Australia Ltd Buy 7.0,105 2.123 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 20,105 3,718 UBS Securities Australia Ltd Buy 26,311 1,320 UBS Securities Australia Ltd Buy 2,629 3,718 UBS Securities Australia Ltd Buy	09-Jul-14	UBS Securities Australia Ltd	BUY	6,575	969	Ordinary
UBS Securities Australia Ltd Buy 5,165 546 UBS Securities Australia Ltd Buy 5,165 379 UBS Securities Australia Ltd Buy 42,880 4,528 UBS Securities Australia Ltd Buy 42,880 4,528 UBS Securities Australia Ltd Buy 8,229 889 UBS Securities Australia Ltd Buy 109,218 11,533 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 66,406 70,687 UBS Securities Australia Ltd Buy 92,181 97,34 UBS Securities Australia Ltd Buy 66,406 70,687 UBS Securities Australia Ltd Buy 5,209 3,178 UBS Securities Australia Ltd Buy 26,71 2,848 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd	09-Jul-14	UBS Securities Australia Ltd	Buy	18,267	1,931	Ordinary
UBS Securities Australia Ltd Buy 5,165 546 UBS Securities Australia Ltd Buy 3,885 379 UBS Securities Australia Ltd Buy 42,880 4,588 UBS Securities Australia Ltd Buy 36,205 3,897 UBS Securities Australia Ltd Buy 78,412 82,80 UBS Securities Australia Ltd Buy 78,412 8,280 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 2,123 430 UBS Securities Australia Ltd Buy 2,105 2,123 UBS Securities Australia Ltd Buy 2,6105 2,133 UBS Securities Australia Ltd Buy 26,91 2,848 UBS Securities Australia Ltd Buy 26,97 2,848 UBS Securities Australia Ltd Buy 3,15 2,848 UBS Securities Australia Ltd Buy 3,15 4,07 UBS Securities Australia Ltd Buy </td <td>09-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>5.165</td> <td>546</td> <td>Ordinary</td>	09-Jul-14	UBS Securities Australia Ltd	Buy	5.165	546	Ordinary
UBS Securities Australia Ltd Buy 3,885 379 UBS Securities Australia Ltd Buy 42,880 4,528 UBS Securities Australia Ltd Buy 36,905 3,897 UBS Securities Australia Ltd Buy 109,218 11,533 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 6,072 2,181 UBS Securities Australia Ltd Buy 92,181 9,734 UBS Securities Australia Ltd Buy 6,072 2,183 UBS Securities Australia Ltd Buy 35,209 3,718 UBS Securities Australia Ltd Buy 5,09 3,718 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 4,075 4,075 UBS Securities Australia Ltd Buy 4,072 4,075 UBS Securities Australia Ltd <t< td=""><td>09-Jul-14</td><td>IUBS Securities Australia Ltd</td><td>Suv</td><td>5 165</td><td>546</td><td>Ordinary</td></t<>	09-Jul-14	IUBS Securities Australia Ltd	Suv	5 165	546	Ordinary
UBS Securities Australia Ltd Buy 42,880 4,528 UBS Securities Australia Ltd Buy 8,229 889 UBS Securities Australia Ltd Buy 109,218 11,533 UBS Securities Australia Ltd Buy 78,412 8,280 UBS Securities Australia Ltd Buy 78,412 8,280 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,105 2,133 UBS Securities Australia Ltd Buy 32,181 9,734 UBS Securities Australia Ltd Buy 669,406 70,687 UBS Securities Australia Ltd Buy 26,911 2,848 UBS Securities Australia Ltd Buy 26,911 2,848 UBS Securities Australia Ltd Buy 26,911 2,848 UBS Securities Australia Ltd Buy 2,691 2,848 UBS Securities Australia Ltd Buy 2,691 2,648 UBS Securities Australia Ltd	09-Jul-14	UBS Securities Australia Ltd	Birv	3 585	370	Ordinary
UBS Securities Australia Ltd Buy 35,005 3,897 UBS Securities Australia Ltd Buy 35,005 3,897 UBS Securities Australia Ltd Buy 109,218 11,533 UBS Securities Australia Ltd Buy 78,412 4,200 UBS Securities Australia Ltd Buy 78,412 4,002 UBS Securities Australia Ltd Buy 4,072 4,072 UBS Securities Australia Ltd Buy 4,072 4,072 UBS Securities Australia Ltd Buy 20,183 2,181 UBS Securities Australia Ltd Buy 20,187 3,718 UBS Securities Australia Ltd Buy 26,571 2,848 UBS Securities Australia Ltd Buy 31,686 3,147 UBS Securities Australia Ltd Buy 2,2499 3,115 UBS Securities Australia Ltd	00-1-1-14	I IRC Societies Australia 143	0.00	000 CF	6,5	o digital y
UBS Securities Australia Ltd Buy 8,5,905 3,897 UBS Securities Australia Ltd Buy 109,218 11,533 UBS Securities Australia Ltd Buy 109,218 11,533 UBS Securities Australia Ltd Buy 1,932 1,260 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,102 430 UBS Securities Australia Ltd Buy 92,181 9,734 UBS Securities Australia Ltd Buy 92,181 9,734 UBS Securities Australia Ltd Buy 92,181 9,734 UBS Securities Australia Ltd Buy 35,209 3,718 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 36,073 4,075 UBS Securities Australia Ltd Buy 3,696 3,147 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 3,156 3,175 UBS Securities Australia Ltd	1 1 1 00		, and	42,980	4,520	Oldinary
UBS Securities Australia Ltd Bluy 108.229 UBS Securities Australia Ltd Buy 109.218 11.583 UBS Securities Australia Ltd Buy 78,412 8.280 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 20,105 2,132 UBS Securities Australia Ltd Buy 26,201 2,184 UBS Securities Australia Ltd Buy 26,371 1,300 UBS Securities Australia Ltd Buy 26,371 1,300 UBS Securities Australia Ltd Buy 26,371 2,848 UBS Securities Australia Ltd Buy 3,156 3,347 UBS Securities Australia Ltd Buy 3,696 3,347 UBS Securities Australia Ltd Buy 3,696 3,347 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy	09-101-14	UBS Securities Australia Ltd	Ruy	36,905	3,897	Ordinary
UBS Securities Australia Ltd Bby 109,218 11,533 UBS Securities Australia Ltd Buy 78,412 8,200 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,105 2,131 UBS Securities Australia Ltd Buy 92,181 9,724 UBS Securities Australia Ltd Buy 56,2406 70,687 UBS Securities Australia Ltd Buy 56,971 2,848 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 31,696 3,347 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 4,025 4,25 UBS Securities Australia Ltd Buy 22,439 3,115 UBS Securities Australia Ltd Buy 22,439 3,115 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd	09-701-14	UBS Securities Australia Ltd	Buy	8,229		Ordinary
UBS Securities Australia Ltd Bby 78,412 8,280 UBS Securities Australia Ltd Buy 1,932 1,260 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 92,181 9,734 UBS Securities Australia Ltd Buy 55,209 3,718 UBS Securities Australia Ltd Buy 26,971 1,2848 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 3,718 3,718 UBS Securities Australia Ltd Buy 3,159 3,115 UBS Securities Australia Ltd Buy 3,169 3,115 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd	09-301-14	UBS Securities Australia Ltd	Buy	109,218		Ordinary
UBS Securities Australia Ltd Bby 11,932 1,260 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 669,406 3,734 UBS Securities Australia Ltd Buy 12,311 1,300 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 4,025 4,25 UBS Securities Australia Ltd Buy 4,022 4,25 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy	09-Jul-14	UBS Securities Australia Ltd	Buy	78,412		Ordinary
UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,105 2,133 UBS Securities Australia Ltd Buy 92,181 9,734 UBS Securities Australia Ltd Buy 55,209 3,718 UBS Securities Australia Ltd Buy 35,209 3,718 UBS Securities Australia Ltd Buy 36,971 2,848 UBS Securities Australia Ltd Buy 36,971 2,848 UBS Securities Australia Ltd Buy 31,696 3,347 UBS Securities Australia Ltd Buy 31,696 3,347 UBS Securities Australia Ltd Buy 31,696 3,347 UBS Securities Australia Ltd Buy 3,696 3,479 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd		UBS Securities Australia Ltd	Buy	11,932		Ordinary
UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 92,18 9,78 UBS Securities Australia Ltd Buy 669,406 70,687 UBS Securities Australia Ltd Buy 25,209 3,718 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 4,025 4,025 UBS Securities Australia Ltd Buy 4,025 4,025 UBS Securities Australia Ltd Buy 3,115 1,15 UBS Securities Australia Ltd Buy 3,147 1,28 UBS Securities Australia Ltd Buy 3,15 4,025 UBS Securities Australia Ltd Buy 1,288 136 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Bu		UBS Securities Australia Ltd	Buv	4 072		Ordinary
UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 20,105 2,123 UBS Securities Australia Ltd Buy 66,3406 70,687 UBS Securities Australia Ltd Buy 12,311 1,300 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 4,025 4,25 UBS Securities Australia Ltd Buy 4,025 4,25 UBS Securities Australia Ltd Buy 4,025 3,347 UBS Securities Australia Ltd Buy 4,025 3,347 UBS Securities Australia Ltd Buy 4,025 3,475 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd	ı	ILIBS Sequities Australia Ltd	Bury	C70 A		(Bulleton)
UBS Securities Australia Ltd Buy 92,110 UBS Securities Australia Ltd Buy 92,181 9,734 UBS Securities Australia Ltd Buy 92,181 9,734 UBS Securities Australia Ltd Buy 35,209 3,718 UBS Securities Australia Ltd Buy 76,971 2,848 UBS Securities Australia Ltd Buy 76,971 2,848 UBS Securities Australia Ltd Buy 4,725 4,725 UBS Securities Australia Ltd Buy 3,159 3,115 UBS Securities Australia Ltd Buy 3,115 3,115 UBS Securities Australia Ltd Buy 1,28,499 3,115 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 4,072 4,90 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy	ı	I IRS Securities Australia I td	Bury	30100		y align
UBS Securities Australia Ltd Buy 56,209 UBS Securities Australia Ltd Buy 669,406 UBS Securities Australia Ltd Buy 12,311 1,300 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 4,025 4,25 UBS Securities Australia Ltd Buy 4,025 4,25 UBS Securities Australia Ltd Buy 31,696 3,347 UBS Securities Australia Ltd Buy 29,499 3,115 UBS Securities Australia Ltd Buy 1,288 1,15 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy 4,072 4	1	100 00 00 00 00 00 00 00 00 00 00 00 00	900	50, 103	2,12	Guirary
UBS Securities Australia Ltd Buy DEBGAMS 70.687 UBS Securities Australia Ltd Buy 35.009 3.718 UBS Securities Australia Ltd Buy 26.971 2.848 UBS Securities Australia Ltd Buy 26.971 2.848 UBS Securities Australia Ltd Buy 4.025 4.25 UBS Securities Australia Ltd Buy 31.696 3,347 UBS Securities Australia Ltd Buy 3.115 UBS Securities Australia Ltd Buy 4,072 4.90 UBS Securities Australia Ltd Buy 4,072 4.29 UBS Securities Australia Ltd Buy 4,072 4.29 UBS Securities Australia Ltd Buy 4,072 4.29	09-Jul-14	UDS Securities Australia Liu	Buy	92,181	9,734	Ordinary
UBS Securities Australia Ltd Buy 3,509 3,718 UBS Securities Australia Ltd Buy 12,311 1,300 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 4,75 5 UBS Securities Australia Ltd Buy 4,025 4,25 UBS Securities Australia Ltd Buy 3,115 3,115 UBS Securities Australia Ltd Buy 1,288 136 UBS Securities Australia Ltd Buy 1,288 136 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,30 UBS Securities Australia Ltd Buy 4,072 4,29 UBS Securities Australia Ltd Buy 4,072 4,29	03-30I-14	ODS SECULINES AUSTRAIN LLU	euy	669,406	/0,68/	Ordinary
UBS Securities Australia Ltd Buy 1,300 UBS Securities Australia Ltd Buy 26,971 2,848 UBS Securities Australia Ltd Buy 4,025 425 UBS Securities Australia Ltd Buy 4,025 425 UBS Securities Australia Ltd Buy 2,9489 3,347 UBS Securities Australia Ltd Buy 1,288 136 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd Buy 4,072 4,290	09-JUI-14	UBS Securities Australia Ltd	Buy	35,209	3,718	Ordinary
UBS Securities Australia Ltd Bby 26,971 2,848 UBS Securities Australia Ltd Buy 4,75 5 UBS Securities Australia Ltd Buy 4,025 4,25 UBS Securities Australia Ltd Buy 3,1696 3,347 UBS Securities Australia Ltd Buy 1,284 3,115 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 420 UBS Securities Australia Ltd Buy 4,072 420 UBS Securities Australia Ltd Buy 4,052 4,290 UBS Securities Australia Ltd Buy 4,052 4,290	09-Jul-14	UBS Securities Australia Ltd	Buy	12,311	1,300	Ordinary
UBS Securities Australia Ltd Buy 47 5 UBS Securities Australia Ltd Buy 4,025 425 UBS Securities Australia Ltd Buy 31,696 3,347 UBS Securities Australia Ltd Buy 22,499 3,115 UBS Securities Australia Ltd Buy 1,288 136 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,026 4,290 UBS Securities Australia Ltd Buy 4,026 4,290	09-Jul-14	UBS Securities Australia Ltd	Buy	26,971	2,848	Ordinary
USS Securities Australia Ltd Buy 4,025 425 UBS Securities Australia Ltd Buy 31,696 33,47 UBS Securities Australia Ltd Buy 29,499 31,15 UBS Securities Australia Ltd Buy 1,288 136 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd Buy 4,026 4,290 UBS Securities Australia Ltd Buy 4,026 4,290	09-Jul-14	UBS Securities Australia Etd	Buy	47	5	Ordinary
UBS Securities Australia Ltd Buy 31,696 3,347 UBS Securities Australia Ltd Buy 29,499 3,115 UBS Securities Australia Ltd Buy 1,286 136 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd Buy 4,0526 4,290	09-Jul-14	UBS Securities Australia Ltd	Buy	4,025	425	Ordinary
UBS Securities Australia Ltd Buy 29,499 3,115 UBS Securities Australia Ltd Buy 1,288 136 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,026 4,290 UBS Securities Australia Ltd Buy 4,026 4,290	09-Jul-14	UBS Securities Australia Ltd	Buy	31,696	3,347	Ordinary
UBS Securities Australia Ltd Buy 1,288 136 UBS Securities Australia Ltd Buy 4,072 430 UBS Securities Australia Ltd Buy 4,072 4,290 UBS Securities Australia Ltd Buy 4,0526 4,290 UBS Securities Australia Ltd Buy 4,0526 4,290	09-Jul-14	UBS Securities Australia Ltd	Buv	29.499	3 115	Ordinary
UES Securities Australia Ltd Buy 4,072 430 UES Securities Australia Ltd Buy 4,0526 4,290 UES Securities Australia Ltd Buy 4,0526 4,290 UES Securities Australia Ltd Buy 4,0526 4,290 UES Securities Australia Ltd Buy 4,0526 4,290 UES Securities Australia Ltd UES Securi	29-111-14	UBS Securities Australia I td	Rin	1 788	136	Ordinan
UBS Securities Australia UBS Securities Aust	7a-1:1-14	II IBC Con ritine Australia td	10.17	2 L D A	130	Cicinary
UBS Securities Australia Ltd Buy 40,626 4,290	1.111	UDS Securities Australia Liv	fna	2/0,4	700	Ordinary
LIBS Committee Australia 14d D.m.	09-Jul-14	UBS Securities Australia Ltd	Buy	40,626	4,290	Ordinary
1085 SeCurities Australia Ltd A 720 L	09-Jul-14	II IRS Securities Australia Ltd	Riny	809 77	A 720	Ordinary

8 by 1,2864 4,28 4,29 1 By 1,2864 1,29 1,1 1 By 1,2864 1,50 1,20 1 By 1,2864 1,50 1,20 1 By 1,2864 1,20 1,20 1 By 1,2864 1,20 1,20 1 By 1,428 3,54 46 1 By 1,428 3,54 46 1 By 1,2864 1,20 1,20 1 By 1,428 1,50 1,20 1 By 1,434 4,24 46 1 By 1,286 1,120 1,120 1 By 1,130 1,120 1,120 1 By 1,130 1,130 1,130 1 By 1,130 1,130	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	dinary	Ordinary	Ordinary	Ordinary	Ordinary	dinary	Ordinary	Ordinary	dinary	dinary	Ordinary	dinany	Ordinary	Ordinary	Ordinary	dinary	Ordinary	Ordinary	dinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	dinary	Ordinary	Ordinary	Ordinary	dinary	dinary	Ordinary	Ordinary	Ordinary	linary	Ordinary	Ordinary	Ordinary	Jinary	dinary	dinary	Ordinary	Ordinary	linary	Ordinary
Buy Buy			13													1	↓	5 6	L													L			5													╛	_		2 V	L	L	Ц	ŏ	1				
	4,068	2,384	123	6	15,864	12,364	11,409	33,498	492	454	2,242	435	130,340	n :	1,864	15,9/2	0	0	16,394	137,643	11,920	123	2,129	20,765	44.734	1,561	19	4,484	4,475	0 00	2,958	47,404	2,724	445	99	7 838	6	1,428	909	95,035	44,944	4 475	13,433	22,411	8,959	4,068	287	61,310	19,119	206,905	99	473	28	445	6	6 6	3 045	610,2	400,0	2,242
titles Australia Ltd titles Australia Ltd	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Bury	and a	Buy	Buy	Buy	Buy	Buy	8ny	Buy	R ₁₃	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Biry	Buy	Buy	Buy	Buy	Buy	Riv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Ruy	Buy	Buy	Buy	Buy	Ang.	Buy	Buy B	Buy	Buy
4[위] [위	Intres Australia Ltd	urities Australia Ltd	urties Australia Ltd	urities Australia Ltd	urities Australia Ltd	urties Australia Ltd	ırities Australia Ltd	urities Australia Ltd	urities Australia Ltd	ırities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	irilles Australia Ltd	intes Australia Ltd	rities Australia Ltd	rrities Australia itd	rities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	inties Australia Ltd	rrities Australia Ltd	irities Australia Ltd	infies Australia Ltd	rities Australia Ltd	rities Australia Ltd	rities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	intres Australia Ltd	rities Australia Ltd	rities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	irties Australia Ltd	rities Australia Ltd	rities Australia Ltd	UBS Securities Australia Ltd	rities Australia Ltd	rities Australia Ltd	rities Australia Ltd	rities Australia 11d	rities Australia Ltd	rities Australia Ltd	Securities Australia Ltd	rities Australia Ltd	rities Australia Ltd	rries Australia Ltd	rrites Australia Ltd	rities Australia Ltd	rities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	rities Australia Ltd	rities Australia Ltd	rities Australia Ltd	rities Australia Ltd	rilles Ausuland Liu	rities Australia Ltd

- Inc-Co					
-Jul-14	UBS Securities Australia Ltd	Buy	38,356	4,029	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	4,094	430	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	5,998	630	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	866'5	069	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	38,356	4.029	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	4,094	430	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	95,896	9.758	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	112.393	11.806	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Briv	48	5	Ordinan
09-111-14	LIBS Securities Australia (1d	Bric	080 85	100	Coming
09-11-d	I IBS Socienties Australia 14d	600	200,000	170,4	Cidinary
11 10	I IBC Committee Australia Ltd	Ang	/98'77	2,402	Ordinary
-101-14	Opp Securities Australia Ltd	Buy	14,908	1,566	Ordinary
09-70I-14	UBS Securities Australia Ltd	Buy	51,932	5,455	Ordinary
09-JUI-14	UBS Securities Australia Ltd	Buy	10,063	1,057	Ordinary
2-JUI-14	UBS Securities Australia Ltd	Buy	113,973	11,972	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	7,064	742	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	11,995	1,260	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	3,560	374	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	40,203	4,223	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	35,491	3,728	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	52,886	2,404	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	7,597	798	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	140,648	14,774	Ordinary
F-Jul-14	UBS Securities Australia Ltd	Buy	1,228	129	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	4,798	204	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	17,955	1,886	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	32,073	3,369	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	27,570	2.896	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	1361	143	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	3.284	345	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	999	02	Ordinan
39-Jul-14	UBS Securities Australia ! td	Birv	1571	165	ordinary ordinary
39-Jul-14	UBS Securities Australia 11d	Biv	8 111	857	Colling
09-1-11-14	1.88 Securities Australia 11d	Birk	25036	000 0	o diam's
11 17	Libe Consider Assessment Liberton	, and	266,03	670'7	Ordinary
09-1-1-17	I IRC Committee Australia 1+d	903	240	97	Ordinary
11 14	Libe comities Australia Lib	Buy	756,1	507	Ordinary
20-101-14	UBS Securities Australia Ltd	BUY	018,84	6,125	Ordinary
- 10	Obs securities Australia Ltd	Buy	1,1,1	123	Ordinary
29-Jul-14	UBS Securities Australia Ltd	Buy	93,858	658'6	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	4,094	430	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	11,995	1,260	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	216,666	22,759	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	52	е	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	248	26	Ordinary
09-}⊓{-60	UBS Securities Australia Ltd	Buy	10	-	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	10	-	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	1,580	166	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	98	6	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	29,417	3.090	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	9,444	266	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	19 640	2,063	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	910	96	Ordinary
99-Jul-14	UBS Securities Australia I td	Buy	12 566	1 320	Ordinary
09-Jul-14	UBS Securities Australia Itd	Birv	26/27	9295	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	2 456	258	Ordinary
-111-14	ILIBS Securities Australia Itd	Bar	71C A	270	Gdinary
09-11-14	LIBS Securities Australia Ltd	Buy	17 215	1 015	Cracialy
11 100	LIBS Commission Australia Ltd	Son d	616,71	518,1	Ordinary
	UBS Securities Australia Ltd	Buy	12,850	1,348	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	42,653	4,471	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	36,968	100	-
				10,0,0	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	46,698	4.895	Ordinary

		a	4.072	430	
09-Jul-14	UBS Securities Australia Ltd	oay		-	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	52,985	5,595	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	4,072	430	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	4,072	430	Ordinary
1-14	UBS Securities Australia Ltd	Buv	62.180	6.566	Ordinary
09-1-1-14	I BS Securities Australia 14d	Bix	207 14	A 730	Ordinan,
29-lul-14	UBS Securities Australia I td	Buy	29 793	4 202	Ordinan
79-1-4-14	I IBS Securities Australia I to	D C	737.63	7,502	Cadinar
1-14	I IBS Securities Australia I td	Bury	13 504	1 426	Ordinati
09-1-1-14	II IRS Securities Australia I to	8.2	100,01	C527	Ordinary
11-00	I IRC Countries Australia 14d	600	300 0	210,	Ordinary
70.101.00	LIBS Securities Australia Ltd	and a	2,300	1 005	Ordinary
20 14 14	I IBC Contraction Australia Ltd	200	10,033	C44.2	Ordinary
	I ID Counting Australia 144	,50	10,100	741.0	Ordinally Ordinally
09-1-1-1-4	I BS Contribe Australia Ltd	200	2000 0	200	Ordinary
20 14 14	1 IDC Contrition Australia Ltd	in o	4 385	300	Commany
11 14 00	1 DC Complete Australia Ltd	Sino d	4,303	403	Ordinary
20 1.1 14	Upo con mitter Australia Ltd	(na	4 000 .	0	Orolinary
70-10-17	I IBC Cod wities Australia 114	Series and a serie	000,1	4 000	Ordinary
11.14	I IRS Securities Australia Ltd	200	36,38	0003	Ordinary
177	I IBC Securities Australia 14d	200	30,00	021.0	Ordinary
09-1-1-14	I IBS Securities Australia I to	Bin	1 205	1001	Ordinary.
0.1.1.1	I IBS Securities Australia 14d	6	ACT 3	710	Ordinary
79-11-14	I IBS Securities Australia I td	200	710.01	1 100	Ordinary
10.00	I IRC Constitute Australia 144	, and	7157	1,100	Ordinary
9-1-1-12	I IRS Securities Australia 14d	Bin	1881	000	Ordinary
1-14	I IRS Securities Australia 14d	i i i	5,71	200	Ordinary
09-1-1-14	URS Securities Australia 14d	6.8	200	2,5	Ordinary
0.1.1.4	I IBS Securities Australia I to	Bin	920	121	Ordinary California
00-1-1-1	I IBC Contrition Australia 14d	600	570 1	007	Ordinary
10-1-14	I IBS Societies Australia 14d	62.0	1,0,7	272	Colling
D 1-1-14	I IBC Countitios Australia 14d	200	000	5	Ordinary
-	I IBC Convities Australia Ltd	600	07	250	Ordinary
03-Jul-14	Cos seculties Australia Ltd	Buy	065,5	000	Ordinary
<u>.</u>	Ubb Securities Australia Ltd	any	07/'07	2,181	Ordinary
9-JUI-14	UBS Securities Australia Ltd	gnà	010,61	7,001	Ordinary
29-JUI-14	UBS Securities Australia Ltd	Buy	3,610	380	Ordinary
09-Jui-14	UBS Securities Australia Ltd	Buy	3,933	414	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	31,607	3,327	Ordinary
1-14	UBS Securities Australia Ltd	Buy	1,216	128	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	864,928	91,045	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	18,582	1,956	Ordinary
J9-Jul-14	UBS Securities Australia Ltd	Buy	3,164	333	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	5,187	546	Ordinary
1-14	UBS Securities Australia Ltd	Buv	9.358	985	Ordinary
09-list-14	LIBS Securities Australia Ltd	Bir	5,000	624	Ordinary
	I IDO Cocuriétos Australia Led	623	7 102	170	Ordinary
- Jul-14	Upo Securities Australia Ltd	Buy	7,18/	240	Ordinary
J9-JUI-14	UBS Securities Australia Ltd	Buy	876'5	624	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	296,372	31,197	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	129,846	13,668	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	9,158	964	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	15,941	1,678	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buv	11.970	1.260	Ordinary
94-111-14	LIBS Sequrities Australia 14d	Biss	15 822	1 662	Ordinan
	P. DC Consister Australia 1+d		220,01	200,-	Colonialy
1	FIDS Countries Australia Ltd	in a	000.00	2,402	Ordinary
#1-Int-60	Coo Securities Australia Liu	bay	560,00	3,101	Ordinary
: :	Con Securities Australia Liu	Buy	40 6	8 1	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	3,342	351	Ordinary
	Cop Securities Australia Liu	ong.	800,111	800,1	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	4,332	455	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	62,270	6,541	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	14,937	1,569	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	3,208	337	Ordinary
)9-Ju -14	UBS Securities Australia Ltd	Buy	857	90	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buv	25	2	-
				2	Ordinary
7	IJBS Securities Australia 11d	Birv	457	48	Ordinary

CONTAIN CONTAIN STATE TAY TO CONTROL CONTAIN OSS SCRUTTER ALBERTINALING BAY 7,775 7,77 7,77 CONTAIN OSS SCRUTTER ALBERTINALING BAY 7,77 7,77 7,77 CONTAIN OSS SCRUTTER ALBERTINALING BAY 7,77 7,77 7,27 CONTAIN OSS SCRUTTER ALBERTINALING BAY 7,77 7,27 7,27 CONTAIN OSS SCRUTTER ALBERTINALING BAY 7,77 7,27 7,27 CONTAIN OSS SCRUTTER ALBERTINALING BAY 7,77 7,27 7,27 CONTAIN OSS SCRUTTER ALBERTING BAY 7,77 7,77 7,77	19-Jul-14			4,,,,		,
(45) Securities Australia Ltd Buy 7,726 8,14 (45) Securities Australia Ltd Buy 7,726 8,14 (45) Securities Australia Ltd Buy 7,17 7,17 (45) Securities Australia Ltd Buy 7,17 7,17 (45) Securities Australia Ltd Buy 7,17 7,17 (46) Securities Australia Ltd Buy 7,17 7,17 (47) Securities Australia Ltd Buy 7,27 8,17 (48) Securities Australia Ltd Buy 7,27 8,17 (48) Securities Australia Ltd Buy 7,27 8,27 (48) Securities Australia Ltd Buy 7,26 1,27 (48) Securities Australia Ltd Buy 7,20 1,27 (48) Securities Australia Ltd Buy 7,20 1,27 (48) Securities Australia Ltd Buy	11.1.14	Upo Securities Austraria Lto	Buy	5,142	539	Ordinary
(18) Securities Australia Lid 64y (1,985) 2,08 (18) Securities Australia Lid 64y 3,333 3,48 (18) Securities Australia Lid 84y 67 7,7 (18) Securities Australia Lid 84y 7,7 7 (18) Securities Australia Lid 84y 7,7 8,7 (18) Securities Australia Lid 84y 7,7 7 (18) Securities Australia Lid 84y 7,6 7 (18) Securities Australia Lid 84y 8,0 8,0 8,0 (18) Securities Australia Lid 84y 8,0 8,0 8,0 (18) Securities Australia Lid 84y 8,0 8,0 8,0 (18) Securities Australia Lid 84y	11111	1 100 Committee Australia Ltd	and and a	500'7	747	Orginary
URS Securities Australia Ltd Buy 3,349 LOS Scoutines Australia Ltd Buy 3,494 LOS Scoutines Australia Ltd Buy 5,19 5,43 LOS Scoutines Australia Ltd Buy 5,19 5,43 LOS Scoutines Australia Ltd Buy 7,7 6,10 LOS Scoutines Australia Ltd Buy 7,7 7 LOS Scoutines Australia Ltd Buy 2,8 7 LOS Scoutines Australia Ltd Buy 2,8 7 LOS Scoutines Australia Ltd Buy 2,8 7 7 LOS Scoutines Australia Ltd Buy 2,8 7 7 LOS Scoutines Australia Ltd Buy 2,8 7 7 LOS Scoutines Australia Ltd Buy 2,9 7 7		COS Seculties Australia Ltd	ána	90/1/	8 4	Ordinary
UBS Scarrintes Australia Lud Buy 5,180 5,43 UBS Scarrintes Australia Lud Buy 5,79 8,17 UBS Scarrintes Australia Lud Buy 7,79 8,17 UBS Scarrintes Australia Lud Buy 8,29 8,10 UBS Scarrintes Australia Lud Buy 8,29 8,20 UBS Scarrintes Australia Lud Buy 8,20 8,20 UBS Scarrintes Australia Lud Buy 8,20<	-Jul-14	UBS Securities Australia Ltd	BUY	1,965	506	Ordinary
(85 Scorrifes Australia Lud Buy 57 HB 54	100	Upo Securities Australia Ltd	Buy	33,333	3,494	Ordinary
OS Securities Australia Ltd Buy 165 108 Securities Australia Ltd Buy 17,94 817 108 Securities Australia Ltd Buy 17,94 817 108 Securities Australia Ltd Buy 50.99 3,647 108 Securities Australia Ltd Buy 6.99 3,647 108 Securities Australia Ltd Buy 6.99 3,647 108 Securities Australia Ltd Buy 6.92 5.92 108 Securities Australia Ltd Buy 6.92 5.93 108 Securities Australia Ltd Buy 5.94 5.95 108 Securities Australia Ltd Buy 5.94 5.95 108 Securities Australia Ltd Buy 7.400 5.94 108 Securities Australia Ltd Buy 7.400 5.95 108 Securities Australia Ltd Buy 7.75 5.97 108 Securities Australia Ltd Buy 7.75 5.97 108 Securities Australia Ltd Buy 7.75 5.97 108 Securities Australia Ltd Buy 7.75 5.	1	Libe contines Australia Liu	ens	001,0	543	Ordinary
UBS Securities Australia Ltd Buy 7,154 817 10B Securities Australia Ltd Buy 7,74 817 67 10B Securities Australia Ltd Buy 67 7 67 7 10B Securities Australia Ltd Buy 67 7 7 67 7 10B Securities Australia Ltd Buy 67 7 7 67 7 10B Securities Australia Ltd Buy 68 67 7 7 67 7 7 67 7 7 67 7 7 67 7 7 67 7 7 67 7 7 67 7 7 67 67 7 67 67 7 67 67 7 67 67 7 67 67 7 67 67 67 67 67 67 67 67 67 67 67 67 67 67 67 67 67 67 67		LIDS Consider Australia and	Social Control	ò	` ;	Orginary
OS Scorrifes Autorials Ltd Buy 7.79 BI 108 Securities Autorials Ltd Buy 7.79 BI 108 Securities Autorials Ltd Buy 5.62 3.67 108 Securities Autorials Ltd Buy 6.62 3.67 108 Securities Autorials Ltd Buy 6.62 8.70 108 Securities Autorials Ltd Buy 2.84 3.01 108 Securities Autorials Ltd Buy 2.84 3.02 108 Securities Autorials Ltd Buy 2.84 3.02 108 Securities Autorials Ltd Buy 2.84 3.02 108 Securities Autorials Ltd Buy 6.07 3.69 108 Securities Autorials Ltd Buy 6.07 3.69 108 Securities Autorials Ltd Buy 7.75 3.69 108 Securities Autorials Ltd Buy 7.75	100	Upo Complete Australia Ltd	ond	501	= ;	Ordinary
OS Securities Australia Ltd Buy 10 (BS Securities Australia Ltd Buy 203 (BS Securities Australia Ltd Buy 209 (BS Securities Australia Ltd Buy 67 (BS Securities Australia Ltd Buy 630 (BS Securities Australia Ltd Buy 630 (BS Securities Australia Ltd Buy 630 (BS Securities Australia Ltd Buy 5240 (BS Securities Australia Ltd Buy 5230	1	I IDO Coomistor A cottonia Lea	And C	46(')	200	Orginary
UBS Securities Australia Ltd Buy 2,94 UBS Securities Australia Ltd Buy 26,095 3,047 UBS Securities Australia Ltd Buy 6,89 3,047 UBS Securities Australia Ltd Buy 6,89 870 UBS Securities Australia Ltd Buy 6,39 870 UBS Securities Australia Ltd Buy 6,400 2,693 10,107 UBS Securities Australia Ltd Buy 5,894 30,30 UBS Securities Australia Ltd Buy 2,640 1,697 UBS Securities Australia Ltd Buy 2,640 1,697 UBS Securities Australia Ltd Buy 2,694 1,697 UBS Securities Australia Ltd Buy 2,040 2,040 UBS Securities Australia Ltd Buy	11-1-1	I IRC Counties Australia Ind	Sing.	70.7	0	Orginary
USS Securities Australia Liu Buy 2,93 3,047 USS Securities Australia Liu Buy 6,62 5,92 UBS Securities Australia Liu Buy 6,83 5,93 UBS Securities Australia Liu Buy 6,83 5,93 UBS Securities Australia Liu Buy 6,93 5,04 UBS Securities Australia Liu Buy 5,045 6,52 UBS Securities Australia Liu Buy 5,049 7,00 UBS Securities Australia Liu Buy 5,049 7,00 UBS Securities Australia Liu Buy 5,049 7,00 UBS Securities Australia Liu Buy 6,04 6,04 UBS Securities Australia Liu Buy 7,489 7,87 UBS Securities Australia Liu Buy 6,04 6,07 UBS Securities Australia Liu Buy 7,189 9,00 UBS Securities Australia Liu Buy 7,175 5,00 UBS Securities Australia Liu Buy 6,04 6,04 UBS Securities Australia Liu Buy <t< td=""><td>1111</td><td>ti De Constitute Australia Ltd</td><td>ouy b</td><td>0 00</td><td>- [</td><td>Ordinary</td></t<>	1111	ti De Constitute Australia Ltd	ouy b	0 00	- [Ordinary
OBS Securities Australia Ltd Buy 6,20,29 3,47 LBS Securities Australia Ltd Buy 6,882 6,982 LBS Securities Australia Ltd Buy 6,882 6,982 LBS Securities Australia Ltd Buy 6,882 6,982 LBS Securities Australia Ltd Buy 2,840 1,697 LBS Securities Australia Ltd Buy 6,607 4,007 LBS Securities Australia Ltd Buy 6,607 4,007 LBS Securities Australia Ltd Buy 6,607 6,007 LBS Securities Australia Ltd Buy 2,107 2,107 LBS Securities Australia Ltd Buy 2,107 2,107 LBS Securities Australia Ltd	100	Upo Securities Australia Ltd	had	930	/ S	Ordinary
Obs. Securities Australia Ltd Buy 6.9 LBS Securities Australia Ltd Buy 8.309 870 LBS Securities Australia Ltd Buy 5.845 612 LBS Securities Australia Ltd Buy 5.845 612 LBS Securities Australia Ltd Buy 5.845 612 LBS Securities Australia Ltd Buy 2.400 2.555 LBS Securities Australia Ltd Buy 2.400 2.555 LBS Securities Australia Ltd Buy 2.400 2.555 LBS Securities Australia Ltd Buy 2.400 1.657 LBS Securities Australia Ltd Buy 2.386 2.497 LBS Securities Australia Ltd Buy 2.386 2.745 LBS Securities Australia Ltd Buy 2.386 2.745 LBS Securities Australia Ltd Buy 2.386 <td>1</td> <td>Pipe Counting Australia Ltd</td> <td>had</td> <td>29,099</td> <td>3,047</td> <td>Orginary</td>	1	Pipe Counting Australia Ltd	had	29,099	3,047	Orginary
OBS Securities Australia Ltd Buy 8,48 8,70 UBS Securities Australia Ltd Buy 5,682 595 UBS Securities Australia Ltd Buy 5,682 595 UBS Securities Australia Ltd Buy 5,894 595 UBS Securities Australia Ltd Buy 5,894 303 UBS Securities Australia Ltd Buy 2,894 303 UBS Securities Australia Ltd Buy 7,460 1,697 UBS Securities Australia Ltd Buy 7,469 1,697 UBS Securities Australia Ltd Buy 1,697 2,697 UBS Securities Australia Ltd Buy 2,797 1,797 UBS Securities Australia Ltd Buy 2,797 1,797 UBS Securities Australia Ltd Buy 2,797 2,797 UBS Securities Australia Ltd Buy	11 14	Los Securities Australia Ltd	bay	/ο	, ,	Ordinary
UBS Securities Australia Ltd Buy 5,845 870 UBS Securities Australia Ltd Buy 5,845 870 UBS Securities Australia Ltd Buy 5,845 612 UBS Securities Australia Ltd Buy 2,640 2,525 UBS Securities Australia Ltd Buy 2,640 2,525 UBS Securities Australia Ltd Buy 2,640 2,525 UBS Securities Australia Ltd Buy 2,640 1,647 UBS Securities Australia Ltd Buy 2,848 1,647 UBS Securities Australia Ltd Buy 2,899 1,647 UBS Securities Australia Ltd Buy 2,896 3,732 UBS Securities Australia Ltd Buy 4,107 2,697 UBS Securities Australia Ltd Buy 2,750 2,697 UBS Securities Australia Ltd Buy 2,750 2,697 UBS Securities Australia Ltd Buy 2,775 3,006 UBS Securities Australia Ltd Buy 2,775 3,006 UBS Securities Australia Ltd Buy	2	UBS Securities Australia Ltd	Buy	48	2	Ordinary
UBS Securities Australia Ltd Bluy 5,682 5682 UBS Securities Australia Ltd Bluy 5,840 5,687 UBS Securities Australia Ltd Bluy 5,840 2,555 UBS Securities Australia Ltd Bluy 5,894 303 UBS Securities Australia Ltd Bluy 5,894 303 UBS Securities Australia Ltd Bluy 7,669 1,697 UBS Securities Australia Ltd Bluy 7,669 1,697 UBS Securities Australia Ltd Bluy 7,668 1,787 UBS Securities Australia Ltd Bluy 1,697 2,894 3,02 UBS Securities Australia Ltd Bluy 1,697 3,02 1,697 UBS Securities Australia Ltd Bluy 4,107 4,30 1,68 UBS Securities Australia Ltd Bluy 4,107 4,30 1,98 UBS Securities Australia Ltd Bluy 4,107 4,30 1,168 UBS Securities Australia Ltd Bluy 2,58 1,168 UBS Securities Australia Ltd Bluy	+101-14	UBS Securities Australia Ltd	Buy	8,309	870	Ordinary
UBS Securities Australia Ltd Buy 5,845 612 UBS Securities Australia Ltd Buy 24,000 2,555 UBS Securities Australia Ltd Buy 9,999 1,047 UBS Securities Australia Ltd Buy 6,67 78 UBS Securities Australia Ltd Buy 7,68 78 UBS Securities Australia Ltd Buy 7,468 782 UBS Securities Australia Ltd Buy 7,468 782 UBS Securities Australia Ltd Buy 1,10 4,30 UBS Securities Australia Ltd Buy 40,545 1,20 UBS Securities Australia Ltd Buy 40,75 2,89 UBS Securities Australia Ltd Buy 2,89 1,73 UBS Securities Australia Ltd Buy 2,75 5,00 UBS Securities Australia Ltd Buy 2,75 5,00 UBS Securities Australia Ltd Buy 2,75 3,00 UBS Securities Australia Ltd Buy 2,75 3,00 UBS Securities Australia Ltd Buy 2,	-Jul-14	UBS Securities Australia Ltd	Buy	2,682	595	Ordinary
UBS Securities Australia Ltd Buy 2,400 2,555 UBS Securities Australia Ltd Buy 9,995 1,01 UBS Securities Australia Ltd Buy 9,655 1,01 UBS Securities Australia Ltd Buy 7,669 1,697 UBS Securities Australia Ltd Buy 7,668 1,783 UBS Securities Australia Ltd Buy 7,668 1,783 UBS Securities Australia Ltd Buy 4,107 4,30 UBS Securities Australia Ltd Buy 4,107 4,30 UBS Securities Australia Ltd Buy 4,107 4,30 UBS Securities Australia Ltd Buy 2,55,501 2,67,4 UBS Securities Australia Ltd Buy 2,55,501 2,67,4 UBS Securities Australia Ltd Buy 2,75,5 1,75 UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Ltd	Buy	5,845	612	Ordinary
UBS Securities Australia Ltd Buy 9999 1,047 UBS Securities Australia Ltd Buy 9999 1,047 UBS Securities Australia Ltd Buy 667 782 UBS Securities Australia Ltd Buy 7,688 782 UBS Securities Australia Ltd Buy 7,688 782 UBS Securities Australia Ltd Buy 7,688 782 UBS Securities Australia Ltd Buy 40,5445 42,455 UBS Securities Australia Ltd Buy 40,756 1,697 UBS Securities Australia Ltd Buy 4,775 500 UBS Securities Australia Ltd Buy 2,107 4,97 UBS Securities Australia Ltd Buy 2,107 4,97 UBS Securities Australia Ltd Buy 2,104 4,775 500 UBS Securities Australia Ltd Buy 2,104 4,775 500 UBS Securities Australia Ltd Buy 2,104 4,107 4,107 UBS Securities Australia Ltd Buy 2,104 4,107 4,101 <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>24,400</td> <td>2,555</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Buy	24,400	2,555	Ordinary
UBS Securities Australia Ltd Bluy 2,884 303 UBS Securities Australia Ltd Buy 2,884 303 UBS Securities Australia Ltd Buy 16,206 1,697 UBS Securities Australia Ltd Buy 1,62,206 1,697 UBS Securities Australia Ltd Buy 7,488 7,82 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 8,671 5,03 UBS Securities Australia Ltd Buy 8,671 5,03 UBS Securities Australia Ltd Buy 2,75 2,93 UBS Securities Australia Ltd Buy 2,75 2,75 UBS Securities Australia Ltd Buy 2,75 2,75 UBS Securities Australia Ltd Buy	-Jul-14	UBS Securities Australia Ltd	Buy	666'6	1,047	Ordinary
UBS Securities Australia Ltd Bluy 2,884 303 UBS Securities Australia Ltd Buy 6,206 1,697 UBS Securities Australia Ltd Buy 1,6206 1,697 UBS Securities Australia Ltd Buy 1,6206 1,697 UBS Securities Australia Ltd Buy 4,107 4,30 UBS Securities Australia Ltd Buy 8,671 4,20 UBS Securities Australia Ltd Buy 8,671 4,20 UBS Securities Australia Ltd Buy 8,671 5,08 UBS Securities Australia Ltd Buy 2,39 1,78 UBS Securities Australia Ltd Buy 2,39 1,78 UBS Securities Australia Ltd Buy 2,750 2,89 UBS Securities Australia Ltd Buy 2,750 2,130 UBS Securities Australia Ltd Buy	-Jul-14	UBS Securities Australia Ltd	Buy	3962	101	Ordinary
UBS Securities Australia Ltd Buy 16,206 1,697 UBS Securities Australia Ltd Buy 1,6206 1,697 UBS Securities Australia Ltd Buy 7,488 7,88 UBS Securities Australia Ltd Buy 7,488 7,82 UBS Securities Australia Ltd Buy 405,437 42,02 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 8,71 9,08 UBS Securities Australia Ltd Buy 2,55,501 2,674 UBS Securities Australia Ltd Buy 2,55,501 2,674 UBS Securities Australia Ltd Buy 2,756 2,69 UBS Securities Australia Ltd Buy 2,756 2,69 UBS Securities Australia Ltd Buy 2,756 2,69 UBS Securities Australia Ltd Buy 2,760 2,68 UBS Securities Australia Ltd Buy 2,176 2,78 UBS Securities Australia Ltd Buy 2,176 2,78 UBS Securities Australia Ltd <td< td=""><td>-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>2,894</td><td>303</td><td>Ordinary</td></td<>	-Jul-14	UBS Securities Australia Ltd	Buy	2,894	303	Ordinary
UBS Securities Australia Ltd Buy 16,206 1,697 UBS Securities Australia Ltd Buy 7,468 7,82 UBS Securities Australia Ltd Buy 4,107 4,39 4,255 UBS Securities Australia Ltd Buy 67,879 9,202 UBS Securities Australia Ltd Buy 87,879 9,202 UBS Securities Australia Ltd Buy 87,879 2,437 UBS Securities Australia Ltd Buy 23,846 2,437 UBS Securities Australia Ltd Buy 23,846 2,437 UBS Securities Australia Ltd Buy 23,846 2,437 UBS Securities Australia Ltd Buy 2,756 2,69 UBS Securities Australia Ltd Buy 2,776 2,88 UBS Securities Australia Ltd Buy 2,170 2,08 UBS Securities Austra	-Ju∤-14	UBS Securities Australia Ltd	Buy	49	7	Ordinary
UBS Securities Australia Ltd Buy 7,468 782 UBS Securities Australia Ltd Buy 153 16 UBS Securities Australia Ltd Buy 405,478 4205 UBS Securities Australia Ltd Buy 4107 430 UBS Securities Australia Ltd Buy 20,846 3,222 UBS Securities Australia Ltd Buy 20,846 3,222 UBS Securities Australia Ltd Buy 20,5501 26,754 UBS Securities Australia Ltd Buy 25,501 26,754 UBS Securities Australia Ltd Buy 2,756 2,89 UBS Securities Australia Ltd Buy 2,760 2,98 UBS Securities Australia Ltd Buy 2,760 2,98 UBS Securities Australia Ltd Buy 2,760 2,98 UBS Securities Australia Ltd Buy 2,130 2,134 UBS Securities Australia Ltd Buy 2,130 2,134 UBS Securities Australia Ltd Buy 2,130 2,134 UBS Securities Australia Ltd Buy	-Jul-14	UBS Securities Australia Ltd	Buy	16.206	1.697	Ordinary
UBS Securities Australia Ltd Buy 405,445 4255 UBS Securities Australia Ltd Buy 40,5445 42,55 UBS Securities Australia Ltd Buy 41,07 430 UBS Securities Australia Ltd Buy 4,107 439 UBS Securities Australia Ltd Buy 2,2846 2,497 UBS Securities Australia Ltd Buy 2,5911 3,132 UBS Securities Australia Ltd Buy 2,5911 3,132 UBS Securities Australia Ltd Buy 2,5911 3,132 UBS Securities Australia Ltd Buy 2,756 2,891 UBS Securities Australia Ltd Buy 2,756 2,891 UBS Securities Australia Ltd Buy 2,756 2,891 UBS Securities Australia Ltd Buy 1,756 2,736 UBS Securities Australia Ltd Buy 2,736 1,316 UBS Securities Australia Ltd Buy 1,756 2,881 UBS Securities Australia Ltd Buy 2,136 2,136 UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Ltd	Buy	7.468	787	Ordinary
UBS Securities Australia Ltd Bby 405,445 42,455 UBS Securities Australia Ltd Bby 81,879 9,202 UBS Securities Australia Ltd Bby 4,107 430 UBS Securities Australia Ltd Bby 3,282 2,437 UBS Securities Australia Ltd Bby 2,59,11 3,132 UBS Securities Australia Ltd Bby 2,59,11 3,132 UBS Securities Australia Ltd Bby 2,59,11 3,132 UBS Securities Australia Ltd Bby 2,59,11 3,65 UBS Securities Australia Ltd Bby 2,70 1,78 UBS Securities Australia Ltd Bby 2,10 2,70 UBS Securities Australia Ltd Bby 1,70 1,14 UBS Securities Australia Ltd Bby 1,340 1,315 UBS Securities Australia Ltd Bby 1,340 1,315 UBS Securities Australia Ltd Bby 1,340 1,315 UBS Securities Australia Ltd Bby 1,350 1,340 UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Ltd	Buv	153	3,5	Ordinary
UBS Securities Australia Ltd Bby 87,879 9,202 UBS Securities Australia Ltd Bby 4,107 430 UBS Securities Australia Ltd Bby 30,866 3,232 UBS Securities Australia Ltd Bby 23,816 2,497 UBS Securities Australia Ltd Bby 23,911 3,132 UBS Securities Australia Ltd Bby 27,60 26,674 UBS Securities Australia Ltd Bby 2,750 26,671 UBS Securities Australia Ltd Bby 2,750 2,89 UBS Securities Australia Ltd Bby 6,074 6,67 UBS Securities Australia Ltd Bby 1,79 1,78 UBS Securities Australia Ltd Bby 1,79 1,14 UBS Securities Australia Ltd Bby 1,79 1,14 UBS Securities Australia Ltd Bby 1,79 1,14 UBS Securities Australia Ltd Bby 1,594 1,31 UBS Securities Australia Ltd Bby 1,500 1,14 UBS Securities Australia Ltd Bby </td <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>405 445</td> <td>47.455</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Buy	405 445	47.455	Ordinary
UBS Securities Australia Ltd Buy 4,107 4,20 UBS Securities Australia Ltd Buy 30,866 3,23 UBS Securities Australia Ltd Buy 23,846 3,23 UBS Securities Australia Ltd Buy 23,846 3,497 UBS Securities Australia Ltd Buy 25,911 3,132 UBS Securities Australia Ltd Buy 2,752 2,674 UBS Securities Australia Ltd Buy 2,767 2,68 UBS Securities Australia Ltd Buy 2,767 2,68 UBS Securities Australia Ltd Buy 2,715 3,055 UBS Securities Australia Ltd Buy 2,715 3,055 UBS Securities Australia Ltd Buy 1,750 1,14 UBS Securities Australia Ltd Buy 2,136 2,23 UBS Securities Australia Ltd Buy 2,136 2,134 UBS Securities Australia Ltd Buy 2,136 1,414 UBS Securities Australia Ltd Buy 1,526 1,549 UBS Securities Australia Ltd	-Jul-14	UBS Securities Australia Ltd	Buy	87,879	0 202	Ordinary
UBS Securities Australia Ltd Buy 30,866 3,232 UBS Securities Australia Ltd Buy 30,866 3,232 UBS Securities Australia Ltd Buy 25,501 3,673 UBS Securities Australia Ltd Buy 25,501 3,673 UBS Securities Australia Ltd Buy 25,501 3,673 UBS Securities Australia Ltd Buy 25,501 3,68 UBS Securities Australia Ltd Buy 2,760 288 UBS Securities Australia Ltd Buy 2,700 1,799 UBS Securities Australia Ltd Buy 2,710 2,700 UBS Securities Australia Ltd Buy 1,709 1,144 UBS Securities Australia Ltd Buy 2,710 2,130 UBS Securities Australia Ltd Buy 2,130 4,170 UBS Securities Australia Ltd Buy 2,130 1,144 UBS Securities Australia Ltd Buy 4,707 4,301 UBS Securities Australia Ltd Buy 4,707 4,107 UBS Securities Australia Ltd	-111-14	UBS Sequities Australia 11d	2 8	4 107	707/	Ordinary.
USS Securities Australia Ltd Buy 30,866 3,232 USS Securities Australia Ltd Buy 23,846 2,497 USS Securities Australia Ltd Buy 25,911 3,132 USS Securities Australia Ltd Buy 2,775 500 USS Securities Australia Ltd Buy 2,750 2,89 USS Securities Australia Ltd Buy 2,760 2,89 USS Securities Australia Ltd Buy 2,760 2,89 USS Securities Australia Ltd Buy 2,760 2,89 USS Securities Australia Ltd Buy 1,709 1,79 USS Securities Australia Ltd Buy 2,101 2,23 USS Securities Australia Ltd Buy 2,130 2,23 USS Securities Australia Ltd Buy 1,256 2,84 USS Securities Australia Ltd Buy 1,500 1,11 USS Securities Australia Ltd Buy 1,500 1,11 USS Securities Australia Ltd Buy 1,500 1,10 USS Securities Australia Ltd Buy <td>111-14</td> <td>I IRS Securities Australia 114</td> <td>600</td> <td>1,10</td> <td>00.</td> <td>Countries</td>	111-14	I IRS Securities Australia 114	600	1,10	00.	Countries
UBS Securities Australia Ltd Buy 23,846 2,925 UBS Securities Australia Ltd Buy 23,846 2,931 3,132 UBS Securities Australia Ltd Buy 25,551 30,055 30 UBS Securities Australia Ltd Buy 4,775 500 289 UBS Securities Australia Ltd Buy 2,760 289 UBS Securities Australia Ltd Buy 2,760 289 UBS Securities Australia Ltd Buy 2,760 289 UBS Securities Australia Ltd Buy 2,170 2,08 UBS Securities Australia Ltd Buy 2,170 2,08 UBS Securities Australia Ltd Buy 2,170 2,08 UBS Securities Australia Ltd Buy 1,1,54 1,144 UBS Securities Australia Ltd Buy 1,540 1,144 UBS Securities Australia Ltd Buy 1,540 1,144 UBS Securities Australia Ltd Buy 1,500 1,144 UBS Securities Australia Ltd Buy 1,070 1,141 <t< td=""><td>1</td><td>1 IDC Cocurition Australia ted</td><td>600</td><td>01</td><td>- 110</td><td>Cramary</td></t<>	1	1 IDC Cocurition Australia ted	600	01	- 110	Cramary
UBS Securities Australia Ltd Buy 25,501 3,437 UBS Securities Australia Ltd Buy 25,501 3,122 UBS Securities Australia Ltd Buy 255,501 26,754 UBS Securities Australia Ltd Buy 6,074 6,88 UBS Securities Australia Ltd Buy 6,074 6,88 UBS Securities Australia Ltd Buy 2,750 1,79 UBS Securities Australia Ltd Buy 2,101 2,08 UBS Securities Australia Ltd Buy 2,130 1,78 UBS Securities Australia Ltd Buy 4,779 1,71 UBS Securities Australia Ltd Buy 5,549 1,84 UBS Securities Australia Ltd Buy 1,550 1,84 UBS Securities Australia Ltd Buy 1,750 1,84 UBS Securities Australia Ltd Buy<	-inc-	Obs securities Australia Ltd	Buy	30,866	3,232	Ordinary
UBS Securities Australia Ltd Buy 25,931 3,132 UBS Securities Australia Ltd Buy 255,501 26,74 UBS Securities Australia Ltd Buy 275,501 26,74 608 UBS Securities Australia Ltd Buy 6,074 6,68 2,760 1,78 3,055 UBS Securities Australia Ltd Buy 2,760 1,78 3,055 1,055 UBS Securities Australia Ltd Buy 1,794 2,731 2,23 UBS Securities Australia Ltd Buy 1,1,154 1,168 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 2,136 2,84 UBS Securities Australia Ltd Buy 1,2,558 1,316 UBS Securities Australia Ltd Buy 1,2,558 1,414 UBS Securities Australia Ltd Buy 1,070 1,070 UBS Securities Australia Ltd Buy 4,107 1,04 UBS Securities Australia Ltd Buy 1,070 1,04 UBS Securitie	-101-114	UBS Securities Australia Ltd	Buy	23,846	2,497	Ordinary
UBS Securities Australia Ltd BMy 255,501 26,754 UBS Securities Australia Ltd Buy 4,775 908 UBS Securities Australia Ltd Buy 6,074 6.56 UBS Securities Australia Ltd Buy 2,760 289 UBS Securities Australia Ltd Buy 2,175 3,055 UBS Securities Australia Ltd Buy 2,176 1,79 UBS Securities Australia Ltd Buy 2,131 2,23 UBS Securities Australia Ltd Buy 2,136 3,23 UBS Securities Australia Ltd Buy 4,779 1,315 UBS Securities Australia Ltd Buy 4,707 4,306 UBS Securities Australia Ltd Buy 5,549 5,514 UBS Securities Australia Ltd Buy 1,570 1,11 UBS Securities Australia Ltd Buy 5,540 5,58 UBS Securities Australia Ltd Buy 1,500 1,414 UBS Securities Australia Ltd Buy 1,500 1,710 UBS Securities Australia Ltd Buy<	4-14	UBS Securities Australia Ltd	Buy	29,911	3,132	Ordinary
UBS Securities Australia Ltd Buy 4,775 500 UBS Securities Australia Ltd Buy 6,074 6,08 UBS Securities Australia Ltd Buy 5,074 6,08 UBS Securities Australia Ltd Buy 2,760 2,79 1,79 UBS Securities Australia Ltd Buy 1,704 1,70 1,79 UBS Securities Australia Ltd Buy 2,101 2,23 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 1,751 4,376 UBS Securities Australia Ltd Buy 1,570 1,11 UBS Securities Australia Ltd Buy 1,570 1,12 UBS Securities Australia Ltd Buy 1,570 1,12 UBS Securities Australia Ltd Buy 1,570 1,844 UBS Securitie	-Jul-14	UBS Securities Australia Ltd	Buy	255,501	26,754	Ordinary
UBS Securities Australia Ltd Buy 8,671 908 UBS Securities Australia Ltd Buy 6,074 6,08 UBS Securities Australia Ltd Buy 2,760 289 UBS Securities Australia Ltd Buy 2,760 1,799 UBS Securities Australia Ltd Buy 1,709 1,799 UBS Securities Australia Ltd Buy 1,134 2,730 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 4,791 4,376 UBS Securities Australia Ltd Buy 1,549 1,414 UBS Securities Australia Ltd Buy 1,554 1,414 UBS Securities Australia Ltd Buy 1,570 4,10 UBS Securities Australia Ltd Buy 1,070 1,14 UBS Securities Australia Ltd Buy 1,070 1,14 UBS Securities Australia Ltd Buy 1,070 1,84 UBS Securities Australia Ltd Buy	-Jul-14	UBS Securities Australia Ltd	Buy	4,775	200	Ordinary
UBS Securities Australia Ltd Bluy 6,074 6,58 UBS Securities Australia Ltd Buy 2,760 289 UBS Securities Australia Ltd Buy 1,709 1,79 UBS Securities Australia Ltd Buy 1,709 1,79 UBS Securities Australia Ltd Buy 2,101 2,20 UBS Securities Australia Ltd Buy 41,791 4,35 UBS Securities Australia Ltd Buy 41,791 4,315 UBS Securities Australia Ltd Buy 4,107 4,315 UBS Securities Australia Ltd Buy 4,107 4,315 UBS Securities Australia Ltd Buy 5,490 581 UBS Securities Australia Ltd Buy 4,107 4,107 UBS Securities Australia Ltd Buy 4,107 4,101 UBS Securities Australia Ltd Buy	-Jul-14	UBS Securities Australia Ltd	Buy	8,671	806	Ordinary
UBS Securities Australia Ltd Buy 2,760 289 UBS Securities Australia Ltd Buy 1,709 1,799 UBS Securities Australia Ltd Buy 1,709 1,799 UBS Securities Australia Ltd Buy 1,134 1,220 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 4,791 4,376 UBS Securities Australia Ltd Buy 1,2,558 1,315 UBS Securities Australia Ltd Buy 1,3504 1,414 UBS Securities Australia Ltd Buy 1,3504 1,414 UBS Securities Australia Ltd Buy 4,107 4,107 UBS Securities Australia Ltd Buy 1,558 1,414 UBS Securities Australia Ltd Buy 1,500 1,414 UBS Securities Australia Ltd Buy 1,500 1,414 UBS Securities Australia Ltd Buy 1,500 1,414 UBS Securities Australia Ltd <	-Jul-14	UBS Securities Australia Ltd	Buy	6,074	989	Ordinary
UBS Securities Australia Ltd Bby 29,175 3,055 UBS Securities Australia Ltd Bby 1,709 179 UBS Securities Australia Ltd Buy 2,101 2,20 UBS Securities Australia Ltd Buy 2,130 2,233 UBS Securities Australia Ltd Buy 2,130 2,136 UBS Securities Australia Ltd Buy 41,731 4,376 UBS Securities Australia Ltd Buy 43,769 1,881 UBS Securities Australia Ltd Buy 1,894 1,315 UBS Securities Australia Ltd Buy 1,354 1,316 UBS Securities Australia Ltd Buy 1,354 1,107 UBS Securities Australia Ltd Buy 1,070 1,11 UBS Securities Australia Ltd Buy 4,107 4,107 UBS Securities Australia Ltd Buy 1,070 1,104 UBS Securities Australia Ltd Buy 1,070 1,104 UBS Securities Australia Ltd Buy 1,070 1,610 UBS Securities Australia Ltd <t< td=""><td>-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buv</td><td>2 760</td><td>280</td><td>Ordinary</td></t<>	-Jul-14	UBS Securities Australia Ltd	Buv	2 760	280	Ordinary
UBS Securities Australia Ltd Bby 1,709 1,700 1	-Jul-14	UBS Securities Australia Ltd	Buy	29175	3 055	Ordinary
UBS Securities Australia Ltd Buy 15,400 2,088 UBS Securities Australia Ltd Buy 2,101 220 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 4,791 4,376 UBS Securities Australia Ltd Buy 4,675 5,519 UBS Securities Australia Ltd Buy 1,525 1,315 UBS Securities Australia Ltd Buy 1,524 581 UBS Securities Australia Ltd Buy 1,070 410 UBS Securities Australia Ltd Buy 1,070 410 UBS Securities Australia Ltd Buy 1,070 410 UBS Securities Australia Ltd Buy 1,070 4,10 UBS Securities Australia Ltd Buy 1,070 1,21 UBS Securities Australia Ltd Buy 1,070 1,21 UBS Securities Australia Ltd Buy 1,51 1,070 UBS Securities Australia Ltd Buy	111-14	URS Securities Australia 11d	Blix	1 209	170	di di
UBS Securities Australia Ltd Buy 1,154 2,088 UBS Securities Australia Ltd Buy 2,101 2,136 UBS Securities Australia Ltd Buy 2,131 4,376 UBS Securities Australia Ltd Buy 41,791 4,376 UBS Securities Australia Ltd Buy 42,603 5,194 UBS Securities Australia Ltd Buy 5,490 581 UBS Securities Australia Ltd Buy 1,315 88 UBS Securities Australia Ltd Buy 1,350 1,414 UBS Securities Australia Ltd Buy 4,107 4,90 UBS Securities Australia Ltd Buy 1,070 1,11 UBS Securities Australia Ltd Buy 1,070 1,11 UBS Securities Australia Ltd Buy 1,070 1,17 UBS Securities Australia Ltd Buy 1,700 1,204 UBS Securities Australia Ltd Buy 1,710 1,844 UBS Securities Australia Ltd Buy 1,710 1,944 UBS Securities Australia Ltd Buy<	11.17	LIDS Contrains Australia Ltd	in a	607,7	6/1	Ordinary
UBS Securities Australia Ltd Buy 2,101 280 UBS Securities Australia Ltd Buy 2,130 1.23 UBS Securities Australia Ltd Buy 2,130 1.23 UBS Securities Australia Ltd Buy 4,771 4,376 UBS Securities Australia Ltd Buy 1,558 1,315 UBS Securities Australia Ltd Buy 1,570 1,884 UBS Securities Australia Ltd Buy 1,570 1,414 UBS Securities Australia Ltd Buy 1,070 1,17 UBS Securities Australia Ltd Buy 1,070 1,17 UBS Securities Australia Ltd Buy 1,070 1,17 UBS Securities Australia Ltd Buy 1,070 1,070 UBS Securities Australia Ltd Buy 1,070 1,844 UBS Securities Australia Ltd Buy 1,5654 2,034 UBS Securities Australia Ltd Buy 1,5664 1,844 UBS Securities Australia Ltd Buy 1,5664 1,844 UBS Securities Australia Ltd B	1	Obs securities Australia Ltd	hng	19,940	7,088	Ordinary
UBS Securities Australia Ltd Bby 1,154 1,168 UBS Securities Australia Ltd Buy 2,130 2,23 UBS Securities Australia Ltd Buy 41,791 4,376 UBS Securities Australia Ltd Buy 12,558 1,315 UBS Securities Australia Ltd Buy 5,549 581 UBS Securities Australia Ltd Buy 18,947 1,984 UBS Securities Australia Ltd Buy 1,070 1,11 UBS Securities Australia Ltd Buy 4,107 4,30 UBS Securities Australia Ltd Buy 4,107 1,30 UBS Securities Australia Ltd Buy 4,107 1,17 UBS Securities Australia Ltd Buy 1,719 1,80 UBS Securities Australia Ltd Buy 1,719 1,84 UBS Securities Australia Ltd Buy 1,719 1,84 UBS Securities Australia Ltd Buy 1,510 1,84 UBS Securities Australia Ltd Buy 1,510 1,84 UBS Securities Australia Ltd Buy <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>2,101</td> <td>220</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Buy	2,101	220	Ordinary
UBS Securities Australia Ltd Buy 2,130 233 UBS Securities Australia Ltd Buy 41,791 4,376 UBS Securities Australia Ltd Buy 12,558 1,315 UBS Securities Australia Ltd Buy 15,549 1,315 UBS Securities Australia Ltd Buy 15,547 1,984 UBS Securities Australia Ltd Buy 8,480 868 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 1,070 1,12 UBS Securities Australia Ltd Buy 1,070 1,12 UBS Securities Australia Ltd Buy 1,070 1,07 UBS Securities Australia Ltd Buy 1,716 1,844 UBS Securities Australia Ltd Buy 2,034 2,134 UBS Securities Australia Ltd Buy 1,518 1,516 UBS Securities Australia Ltd Buy 1,518 1,516 UBS Securities Australia Ltd Buy 1,516 1,516 UBS Securities Australia Ltd Bu	-Jul-14	UBS Securities Australia Ltd	Buy	11,154	1,168	Ordinary
UBS Securities Australia Ltd Buy 41,791 4,376 UBS Securities Australia Ltd Buy 42,603 5,194 UBS Securities Australia Ltd Buy 12,558 1,315 UBS Securities Australia Ltd Buy 18,947 1,894 UBS Securities Australia Ltd Buy 18,947 1,894 UBS Securities Australia Ltd Buy 1,696 1,89 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 1,719 1,89 UBS Securities Australia Ltd Buy 1,719 1,80 UBS Securities Australia Ltd Buy 1,719 1,84 UBS Securities Australia Ltd Buy 2,034 2,134 UBS Securities Australia Ltd Buy 2,034 2,134 UBS Securities Australia Ltd Buy 2,034 1,510 UBS Securities Australia Ltd Buy 2,034 1,150 UBS Securities Australia Ltd Buy 1,540 1,540 UBS Securities Australia Ltd <t< td=""><td>-Jul-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>2,130</td><td>223</td><td>Ordinary</td></t<>	-Jul-14	UBS Securities Australia Ltd	Buy	2,130	223	Ordinary
UBS Securities Australia Ltd Buy 49,603 5,194 UBS Securities Australia Ltd Buy 1,2558 1,315 UBS Securities Australia Ltd Buy 5,440 581 UBS Securities Australia Ltd Buy 1,640 1,884 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 4,710 180 UBS Securities Australia Ltd Buy 1,719 180 UBS Securities Australia Ltd Buy 1,518 1,518 UBS Securities Australia Ltd Buy 1,518 1,516 UBS Securities Australia Ltd Buy 1,518 1,500 UBS Securities Australia Ltd Buy <td>-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>41 791</td> <td>4 376</td> <td>Ordinary</td>	-Jul-14	UBS Securities Australia Ltd	Buv	41 791	4 376	Ordinary
UBS Securities Australia Ltd Buy 1,2,558 1,315 UBS Securities Australia Ltd Buy 15,549 581 UBS Securities Australia Ltd Buy 18,947 1,944 UBS Securities Australia Ltd Buy 1,070 471 UBS Securities Australia Ltd Buy 4,107 400 UBS Securities Australia Ltd Buy 4,107 40 UBS Securities Australia Ltd Buy 4,107 40 UBS Securities Australia Ltd Buy 4,281 4,710 UBS Securities Australia Ltd Buy 4,481 4,710 UBS Securities Australia Ltd Buy 4,481 4,710 UBS Securities Australia Ltd Buy 1,518 1,51 UBS Securities Australia Ltd Buy 1,514 1,50 UBS Securities Australia Ltd Buy 1,518 1,51 UBS Securities Australia Ltd Buy 1,518 1,510 UBS Securities Australia Ltd Buy 1,648 1,510 UBS Securities Australia Ltd Buy	-Jul-14	UBS Securities Australia Ltd	BIN	49 603	104	Coding
USS Securities Australia Ltd Buy 5,549 5,81 UBS Securities Australia Ltd Buy 18,947 1,984 UBS Securities Australia Ltd Buy 18,947 1,984 UBS Securities Australia Ltd Buy 1,107 1,414 UBS Securities Australia Ltd Buy 1,070 1,12 UBS Securities Australia Ltd Buy 1,719 1,80 UBS Securities Australia Ltd Buy 1,719 1,80 UBS Securities Australia Ltd Buy 4,107 4,107 UBS Securities Australia Ltd Buy 4,4981 4,101 UBS Securities Australia Ltd Buy 2,034 2,134 UBS Securities Australia Ltd Buy 1,510 1,540 UBS Securities Australia Ltd Buy 1,510 1,544 UBS Securities Australia Ltd Buy 1,510 1,544 UBS Securities Australia Ltd Buy 1,510 1,500 UBS Securities Australia Ltd Buy 1,510 1,500 UBS Securities Australia Ltd <t< td=""><td>-111-14</td><td>I IBS Securities Australia Ltd</td><td>Britis</td><td>12 559</td><td>1 215</td><td>O Series</td></t<>	-111-14	I IBS Securities Australia Ltd	Britis	12 559	1 215	O Series
UBS Securities Australia Ltd Buy 18,947 1,984 UBS Securities Australia Ltd Buy 18,947 1,984 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 1,070 112 UBS Securities Australia Ltd Buy 1,070 1,17 UBS Securities Australia Ltd Buy 4,4381 4,710 UBS Securities Australia Ltd Buy 4,4381 4,710 UBS Securities Australia Ltd Buy 1,518 1,844 UBS Securities Australia Ltd Buy 1,518 1,516 UBS Securities Australia Ltd Buy 1,518 1,516 UBS Securities Australia Ltd Buy 1,518 1,516 UBS Securities Australia Ltd Buy 1,518 1,510 UBS Securities Australia Ltd Buy 1,518 1,500 UBS Securities Australia Ltd Buy 1,518 1,500 UBS Securities Australia Ltd	11.14	I IBC Contribute Australia 14d	600	12,339	100	Gulfary
UBS Securities Australia Ltd Buy 1,984 UBS Securities Australia Ltd Buy 1,414 UBS Securities Australia Ltd Buy 1,107 UBS Securities Australia Ltd Buy 1,070 UBS Securities Australia Ltd Buy 1,719 UBS Securities Australia Ltd Buy 1,719 UBS Securities Australia Ltd Buy 1,719 UBS Securities Australia Ltd Buy 2,034 UBS Securities Australia Ltd Buy 2,034 UBS Securities Australia Ltd Buy 1,510 UBS Securities Australia Ltd Buy 1,510 UBS Securities Australia Ltd Buy 1,518 UBS Securities Australia Ltd Buy 1,644 UBS Securities Australia Ltd Buy 1,650 UBS Securities Australia Ltd Buy 1,677 UBS Securities Australia Ltd Buy		Liber Commission Assessing the	503	5500	100	Guinary
UBS Securities Australia Ltd Buy 8,480 888 UBS Securities Australia Ltd Buy 11,504 1,410 13,504 1,410 1,400 1,410 1,400 1,107 1,10		ODS Securities Australia Ltd	ang P	10,347	486,	Orainary
UBS Securities Australia Ltd Buy 1,315.04 1,414 UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 1,670 112 UBS Securities Australia Ltd Buy 1,654 2,088 UBS Securities Australia Ltd Buy 4,710 180 UBS Securities Australia Ltd Buy 4,481 4,710 UBS Securities Australia Ltd Buy 1,518 159 UBS Securities Australia Ltd Buy 1,518 159 UBS Securities Australia Ltd Buy 1,518 1,516 UBS Securities Australia Ltd Buy 1,518 1,159 UBS Securities Australia Ltd Buy 1,518 1,150 UBS Securities Australia Ltd Buy 1,518 1,150 UBS Securities Australia Ltd Buy 1,648 1,115 UBS Securities Australia Ltd Buy 1,630 1,606 UBS Securities Australia Ltd Buy 4,775 500 UBS Securities Australia Ltd Buy <td>100</td> <td>ODD Securities Australia Ltd</td> <td>and a</td> <td>8,480</td> <td>888</td> <td>Ordinary</td>	100	ODD Securities Australia Ltd	and a	8,480	888	Ordinary
UBS Securities Australia Ltd Buy 4,107 430 UBS Securities Australia Ltd Buy 1,070 11,12 UBS Securities Australia Ltd Buy 1,719 180 UBS Securities Australia Ltd Buy 1,719 180 UBS Securities Australia Ltd Buy 2,034 2,13 UBS Securities Australia Ltd Buy 2,034 2,13 UBS Securities Australia Ltd Buy 1,510 1,844 UBS Securities Australia Ltd Buy 1,518 1,150 UBS Securities Australia Ltd Buy 1,645 1,500 UBS Securities Australia Ltd Buy 1,645 1,500 UBS Securities Australia Ltd Buy 1,624 1,600 UBS Securities Australia Ltd Buy 1,627 1,600 UBS Securities Australia Ltd Buy 1,627 1,600 UBS Securities Australia Ltd Buy 1,775 500	-JUI-14	UBS Securities Australia Ltd	Buy	13,504	1,414	Ordinary
UBS Securities Australia Ltd Buy 1,070 112 UBS Securities Australia Ltd Buy 19,654 2,088 UBS Securities Australia Ltd Buy 1,719 180 UBS Securities Australia Ltd Buy 44,981 4,710 UBS Securities Australia Ltd Buy 2,034 2,134 UBS Securities Australia Ltd Buy 1,518 159 UBS Securities Australia Ltd Buy 1,518 159 UBS Securities Australia Ltd Buy 1,644 1,500 UBS Securities Australia Ltd Buy 1,648 1,500 UBS Securities Australia Ltd Buy 1,645 1,500 UBS Securities Australia Ltd Buy 1,645 1,500 UBS Securities Australia Ltd Buy 9,989 1,046 UBS Securities Australia Ltd Buy 9,989 1,046 UBS Securities Australia Ltd Buy 4,775 500	-Jul-14	UBS Securities Australia Ltd	Buy	4,107	430	Ordinary
UBS Securities Australia Ltd Buy 19,654 2,058 UBS Securities Australia Ltd Buy 1,719 180 UBS Securities Australia Ltd Buy 44,981 4,710 UBS Securities Australia Ltd Buy 2,034 2,13 UBS Securities Australia Ltd Buy 1,5610 1,844 UBS Securities Australia Ltd Buy 1,5610 1,844 UBS Securities Australia Ltd Buy 1,610 1,844 UBS Securities Australia Ltd Buy 1,614 1,900 UBS Securities Australia Ltd Buy 16,271 1,066 UBS Securities Australia Ltd Buy 16,371 1,066 UBS Securities Australia Ltd Buy 1,377 1,066 UBS Securities Australia Ltd Buy 4,775 500	-Jul-14	UBS Securities Australia Ltd	Buy	1,070	112	Ordinary
UBS Securities Australia Ltd Buy 1,719 180 UBS Securities Australia Ltd Buy 44,981 47,10 UBS Securities Australia Ltd Buy 2,034 2,13 UBS Securities Australia Ltd Buy 2,034 2,13 UBS Securities Australia Ltd Buy 1,510 1,844 UBS Securities Australia Ltd Buy 1,518 1,15 UBS Securities Australia Ltd Buy 1,8145 1,900 UBS Securities Australia Ltd Buy 1,8145 1,900 UBS Securities Australia Ltd Buy 1,9371 1,086 UBS Securities Australia Ltd Buy 4,775 500	-Jul-14	UBS Securities Australia Ltd	Buy	19,654	2,058	Ordinary
UBS Securities Australia Ltd Buy 44,881 4,710 UBS Securities Australia Ltd Buy 2,034 2,710 UBS Securities Australia Ltd Buy 1,7610 1,844 UBS Securities Australia Ltd Buy 1,518 159 UBS Securities Australia Ltd Buy 10,548 1,15 UBS Securities Australia Ltd Buy 10,548 1,15 UBS Securities Australia Ltd Buy 18,45 1,500 UBS Securities Australia Ltd Buy 10,371 1,066 UBS Securities Australia Ltd Buy 9,989 1,046 UBS Securities Australia Ltd Buy 4,775 500	-Jul-14	UBS Securities Australia Ltd	Buv	1 719	180	Ordinary
UBS Securities Australia Ltd Buy 44,981 4,710 UBS Securities Australia Ltd Buy 2,034 213 UBS Securities Australia Ltd Buy 17,610 1,844 UBS Securities Australia Ltd Buy 1,518 1,915 UBS Securities Australia Ltd Buy 10,648 1,115 UBS Securities Australia Ltd Buy 18,145 1,900 UBS Securities Australia Ltd Buy 10,371 1,086 UBS Securities Australia Ltd Buy 4,775 500	Jul-14	UBS Securities Australia I td	Bliv	38	7	Ordinary
UBS Securities Australia Ltd Buy 1,510	177	I IBC Counties Australia 14d	(-0	44 001		o di ili
UBS Securities Australia Ltd Buy 1,540 1,844 UBS Securities Australia Ltd Buy 1,518 1,98 UBS Securities Australia Ltd Buy 1,518 1,115 UBS Securities Australia Ltd Buy 10,648 1,115 UBS Securities Australia Ltd Buy 16,371 1,080 UBS Securities Australia Ltd Buy 10,371 1,086 UBS Securities Australia Ltd Buy 9,989 1,046 UBS Securities Australia Ltd Buy 4,775 500	: :	De Commission Assistant Life	, and	144,361	01/4	Ordinary
UBS Securities Australia Ltd Buy 1,510 1,844 UBS Securities Australia Ltd Buy 1,518 159 UBS Securities Australia Ltd Buy 10,648 1,15 UBS Securities Australia Ltd Buy 16,371 1,000 UBS Securities Australia Ltd Buy 10,371 1,086 UBS Securities Australia Ltd Buy 4,775 500	101-17	UBS Securities Australia Ltd	Buy	2,034	213	Ordinary
UBS Securities Australia Ltd Buy 1,518 159 UBS Securities Australia Ltd Buy 10,648 1,115 UBS Securities Australia Ltd Buy 18,145 1,500 UBS Securities Australia Ltd Buy 10,371 1,086 UBS Securities Australia Ltd Buy 1,046 1,046 UBS Securities Australia Ltd Buy 4,775 500	-Juf-14	UBS Securities Australia Ltd	Buy	17,610	1,844	Ordinary
UBS Securities Australia Ltd Buy 10,648 1,115 UBS Securities Australia Ltd Buy 18,145 1,900 UBS Securities Australia Ltd Buy 10,371 1,086 UBS Securities Australia Ltd Buy 5,989 1,046 UBS Securities Australia Ltd Buy 4,775 500	-Jul-14	UBS Securities Australia Ltd	Buy	1,518	159	Ordinary
UBS Securities Australia Ltd Buy 18,145 1,900 UBS Securities Australia Ltd Buy 10,2371 1,086 UBS Securities Australia Ltd Buy 9,989 1,046 UBS Securities Australia Ltd Buy 4,775 500	-Jul-14	UBS Securities Australia Ltd	Buy	10,648	1,115	Ordinary
UBS Securities Australia Ltd Buy 10,371 1,086 UBS Securities Australia Ltd Buy 9,989 1,046 UBS Securities Australia Ltd Buy 4,775 500	-Jul-14	UBS Securities Australia Ltd	Buy	18,145	1,900	Ordinary
UBS Securities Australia Ltd Buy 9,989 1,046 UBS Securities Australia Ltd Buy 4,775 500	Jul-14	UBS Securities Australia Ltd	Buy	10.371	1,086	Ordinary
UBS Securities Australia Ltd Buy 4,775 500	-Jul-14	UBS Securities Australia Ltd	Buv	686.6	1 046	Ordinary
USS SECURITIES AUSTRAINE LIG BUY 4,775 500	1.4-17					V 01 10 10 1
	71-11	The second of th	u	4 775	200	

Г	_	_		_		1	_	_	_	_	_	_		_	_		_	_	_		_			_	_	_	1	_			_	1	_	_		_	_	_					_	_,					_		_	_								_	_		_	_	_	_	_
-	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Guinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cidinaly	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary																																			
1010	350	2,970	4,989	430	7,141	3,341	1 063	1 767	70/1	450	1/3	9,855	5,070	1,260	817	2 1	5,146	37	-	5,141	4,097	430	4,892	5,108	5,108	4.067	58	2,201	3.700	430	10.060	4,097	521	1,260	430	546	4,832	4,627	546	430	546	149	096	2,105	3,167	1,153	1,713	2,866	7 55	430	3 257	587	300	19	2,400	-	1,187	3,000	2,731	m	545	430	700	400	3,757	546	1,758
000	3,339	28,334	47,595	4,102	68,125	31,873	10.141	16.809	606,01	4,102	0c9'L	94,017	48,368	12,020	7 794	10000	49,093	353	10	49,045	380'68	4,102	46,670	48.730	48,730	38.799	553	20.998	35,298	4,102	95,972	39,085	4,970	12,020	4,102	5,209	46,097	44,142	5,209	4,102	5,209	1,421	9,158	20,082	30,213	11,000	16,342	2/,342	61	201,5	31,072	5,600	2,862	181	22,896	10	11,324	28,620	26,054	29	5,199	4,102	976,0	3,816	35,842	607'5	16,771
	Ruy	Buy	Buy	Buy	Buy	Buy	NA.	Buy	600	900	Buy	Buy	Buy	Buy	Biv	6.2	Buy	Buv	Buv	Buv	Buv	Buv	Buy	Ruy	any Brit	Riv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	on)	and a	Rug	Buy	Buy																									
7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	Securities	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	JBS Securities Australia ! td	IDC Con mition Ametralia 14d	DOS DECURITION A CANADA A LANGE TO A	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia I to	DO Committee Australia 14d	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	IBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	IBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	85 Securities Australia Ltd	UBS Securities Australia Ltd	105 Securities Australia Lid	BS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	UBS Securities Australia Ltd
Ī		T	1			09-Jul-14	Г	T	T	Ť	Т			09-Jul-14	ı	Ī	Ť	1	7					09-Jul-14	Γ	Γ	Γ	Ť	Ť	ı		09-Jul-14		Ī				09-Jul-14				09-Jul-14			1	1	09-Jul-14	Ť	1	09-Jul-14	Ī	Ī	09-Jul-14						٦	-		1	1	09-101-14	Т	ŀ	09-Jul-14

-14	IOBS Securities Australia Ltd	2.5	4 4P/	882	
09-Jul-14	UBS Securities Australia Ltd	Boy	14.400	1.500	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	42,221	4,398	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	42,163	4,392	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	7,920	825	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	35,069	3,653	Ordinary
4-17	UBS Securities Australia Ltd	Buy	768	80	Ordinary
41-101-14 1-1 -14	UBS Securities Australia Ltd	Buy	4,128	430	Ordinary
09-101-14	I IRC Societies Australia Ltd	Buy	18,528	1,930	Ordinary
9-Jul-14	UBS Securities Australia 14d	Book State	00,230	0,274	Ordinary
ul-14	UBS Securities Australia Ltd	Buck	7.181	748	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	23,386	2.436	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	1,402	146	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	4,128	430	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	16,320	1,700	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buy	74,093	7,718	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	24,413	2,543	Ordinary
Ul- 24	UBS Securities Australia Ltd	Buy	23,050	2,401	Ordinary
4:-17-00	UBS Securities Australia Ltd	Buy	44,525	4,638	Ordinary
29-Jul-14	UBS Securities Australia Ltd	Buy	19,200	2,000	Ordinary
41-11-12	UBS Securities Australia Ltd	Buy	13,651	1,422	Ordinary
41-10-50	UBS Securities Australia Ltd	Buy	1,373	143	Ordinary
29-JUI-14	UBS Securities Australia Ltd	Buy	11,107	1,157	Ordinary
-in-in-	UBS Securities Australia Ltd	Buy	4,128	430	Ordinary
10-10-14	I IBC Countries Australia 14d	boy	16,320	00/	Ordinary
10-11	I IRC Countries Australia 14d	Space of the space	20,750	7000	Ordinary
09-111-14	II RS Securities Australia I td	Box	110 000	3,910	Ordinary
ul-14	UBS Securities Australia Ltd	Suy	4 339	452	Ordinary
9-Jul-14	UBS Securities Australia Ltd	Buv	12.096	1.260	Ordinary
39-Jul-14	UBS Securities Australia Ltd	Buy	4,128	430	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	14,179	1,477	Ordinary
ul-14	UBS Securities Australia Ltd	Buy	12,154	1,266	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	9,293	896	Ordinary
14-lu	UBS Securities Australia Ltd	Buy	58	9	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	16,320	1,700	Ordinary
-14 	UBS Securities Australia Ltd	Buy	4,166	434	Ordinary
29-JUI-14	UBS Securities Australia Ltd	Buy	168,749	17,578	Ordinary
29-701-14	UBS Securities Australia Ltd	Buy	2,842	296	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	43,795	4,562	Ordinary
1 2	UBS Securities Australia Ltd	Buy	20,473	2,115	Ordinary
09-Jul-14	UBS Securities Australia Ltd	Buy	1,220	126	Ordinary
U-14	UBS Securities Australia Ltd	Buy	3,959	409	Ordinary
1-14	UBS Securities Australia Ltd	Buy	13,368	1,381	Ordinary
10-Jul-14	UBS AG London Branch	Buy	525,134	55,193	Ordinary
0-Jul-14	UBS AG London Branch	Sell	524,449	(55,193)	Ordinary
- - - -	UBS AG London Branch	Buy	1,396,794	146,841	Ordinary
10-101-14	UBS AG London Branch	Sell	1,392,042	(146,841)	Ordinary
4	UBS AG London Branch	Sell	598,412	(62,931)	Ordinary
0-101-14	USS AG London Branch	Buy	600,455	62,931	Ordinary
0-701-14	UBS Securities Australia Ltd	Buy	47,228	4,966	Ordinary
41-10	UBS Securities Australia Ltd	Sell	562,021	(59,799)	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	10,050	1,056	Ordinary
1-14	UBS Securities Australia Ltd	Buy	151,213	15,925	Ordinary
10-701-14	UBS Securities Australia Ltd	Seil	342,827	(36,043)	Ordinary
0-101-14	UBS Securities Australia Ltd	Buy	525,134	55,193	Ordinary
P. Inl. 14	1 IBC Contribute Australia Ltd	Sell Dist	390,390	(5/0/14)	Ordinary
0-101-14	1 RS Sequities Australia Ind	body Sell	24,329	5,7/4	Orginary
0-14-14	I IRS Securities Australia 14d	200	576,77	17.544	Ordinary
0-lul-14	188 Securities Australia III	S S	107,902	7,044	Ordinary
0-101-14	LIRS Securities Australia (10	kno Kno	49,525	5,229	Ordinary
10-Jul-14	UBS Securities Australia Ltd	000	120.031	(013 (1)	Ordinary
	510 1010 1010 1010 1010 1010				2000
10-Jul-14	UBS Securities Australia I td	9	358 396	(37 572)	Ordinary

UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy	7,449	780	Ordinary Ordinary Ordinary
Securities Australia Ltd Securities Australia Ltd Securities Australia Ltd	Buy	4,107	430	Ordinary
Securities Australia Ltd Securities Australia Ltd				Ordinary
Securities Australia Ltd	Buy	12,033	1,260	
	Buy	109'06	9,487	Ordinary
UBS Securities Australia Ltd	Buy	8,757	917	Ordinary
JBS Securities Australia Ltd	Buy	5,644	591	Ordinary
UBS Securities Australia Ltd	Виу	11,632	1,218	Ordinary
Securities Australia Ltd	Buy	17,868	1,871	Ordinary
UBS Securities Australia Ltd	Buy	1,996	500	Ordinary
UBS Securities Australia Ltd	Buy	5,911	619	Ordinary
Securities Australia Ltd	Buy	11,995	1,256	Ordinary
UBS Securities Australia Ltd	Buy	3,705	388	Ordinary
5 Securities Australia Ltd	Buy	15,519	1,625	Ordinary
UBS Securities Australia Ltd	Buy	13,418	1,405	Ordinary
UBS Securities Australia Ltd	Buy	134	14	Ordinary
UBS Securities Australia Ltd	Buv	15.691	1 643	Ordinary
IBS Sequrities Australia I to	Bin	14 325	1,500	Ordingo
Cocuration Australia 140	65.0	0701	200,	Ordinary.
ODS Securities Australia Litu	ána d	046,1	3 2 2	Ordinary
Securities Australia Ltd	boy	9/5/77	4,304	Ordinary
JBS Securities Australia Ltd	Buy	4,794	205	Ordinary
5 Securities Australia Ltd	Buy	40,320	4,222	Ordinary
UBS Securities Australia Ltd	Buv	4.107	0.57	Ordinary
Securities Australia 11d	Bun	L 211	246	Ordingo
Libe Countries Assessed and	653	1010	2	o de la company
Securities Australia Lito	ŝna	016,1	007	Grainary
S Securities Australia Liu	, and	42,973	00c,4	Ordinary
UBS Securities Australia Ltd	Buy	11/,436	/67'71	Ordinary
UBS Securities Australia Ltd	Buy	19,702	2,063	Ordinary
UBS Securities Australia Ltd	Buy	7,210	751	Ordinary
Securities Australia Ltd	Buy	8,467	882	Ordinary
Securities Australia Ltd	Buy	7,565	788	Ordinary
IBS Securities Australia Ltd	8th	6778	206	Ordinary
Securities Australia Itd	B. D.	33 763	3 517	Ordinan
IDS Counting America 14d	9.5	55,55	10,00	, in the second
Seculities Australia Liu	(na	2/9/21	026,1	Ordinary
Securities Australia Liu	ana -	06,490	0/1/0	Orginary
UBS Securities Australia Ltd	BUŊ	3,619	377	Ordinary
S Securities Australia Ltd	Buy	28,397	2,958	Ordinary
UBS Securities Australia Ltd	Buy	6,970	726	Ordinary
5 Securities Australia Ltd	Buy	35,472	3,695	Ordinary
UBS Securities Australia Ltd	Buv	4.128	430	Ordinary
IRS Societies Australia Ltd	Bigs	21 907	2 282	Ordinan
Securities Australia Ltd	600	106,12	707'7	Olditaly
s securities Australia Ltd	Buy	21,3/9	7777	Ordinary
Securities Australia Ltd	Buy	22,301	2,323	Ordinary
UBS Securities Australia Ltd	Виу	2,880	300	Ordinary
JBS Securities Australia Ltd	Buy	17,846	1,859	Ordinary
185 Securities Australia Itd	Buy	20 957	7 183	Ordinan
DO Contract A restriction Office	(2)	20,650	2020	Ording
Securities Australia Ltd	ίρα α	000,00	2000,0	Gunaly
UBS Securities Australia LIG	any	40,243	4,81/	Orginary
S Securities Australia Ltd	Buy	49,910	5,199	Ordinary
UBS Securities Australia Ltd	Buy	10,944	1,140	Ordinary
S Securities Australia Ltd	Buv	25.776	2.685	Ordinary
IBS Sequirities Australia I td	Briv	009 88	2 500	Ordinan
IDC Con without Australia 144	Bitt	052 53	8003	Cadina
DO Commission American 14d	0	252.00	1000	1
The District Control of the	, and	D.C.O.	200,5	i i
202 Securities Australia Ltd	, and	h h' h	410,4	Cidinaly
S Securities Australia Ltd	pay	8,141	848	Croinary
JBS Securities Australia Ltd	Buy	24,739	2,577	Ordinary
Securities Australia Ltd	Buy	28,785	3,000	Ordinary
UBS Securities Australia Ltd	Buy	5,242	546	Ordinary
JBS Securities Australia Ltd	Buy	130,992	13,645	Ordinary
JBS Securities Australia Ltd	Buy	4,128	430	Ordinary
IBS Securities Australia (td	Buy	1 920	200	Ordinary
IDC Consider Australia Ind	6.00	25,75	200	O. C.
Securities Australia Ltd	pan	36,/01	5,823	Orginary
JBS Securities Australia Ltd	Buy	1,450	151	Ordinary
S Securities Australia I td	Buy	756 51	1 589	Ordinary
ODD DECURIES MUSICALIA	pay	+C2,C1	500,1	Oldinary

10-Jul-14	UBS Securities Australia Ltd	Sell	257	(22)	Ordinary
10-Jul-14	מס אינונורים שימוומו דונים	S	044 34	(000)	,
7.1	UBS Securities Australia Ltd	E S	07/10	(1000/1)	Orginary
	I BC Securities Australia I+d	1103	024.0	(200)	Ordinary
1 1 1	I IBS Societies Australia Ltd	la d	3,154	(332)	Ordinary
- FI-1-17	I IDC Countition Australia I+4	36	790'7	(777)	Ordinary
0-11-14	I IRS Securities Australia 144	3611	0/6/67	(4,400)	Orginary
0-111-14	I IBS Sequition Australia 14d	Call	000,01	(0+0'1)	Ordinary
hi-14	UBS Securities Australia Ltd	3	3 173	(410)	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Sell	3.164	(333)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	38	24.900	(2 621)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	les.	2.053	(216)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Sell	1,739	(183)	Ordinary
lul-14	UBS Securities Australia Ltd	Sell	3,230	(340)	Ordinary
10-Ju{-14	UBS Securities Australia Ltd	Sell	3,230	(340)	Ordinary
lui-14	UBS Securities Australia Ltd	Sell	1,254	(132)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Sell	5,273	(555)	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Sell	4,731	(498)	Ordinary
Jul-14	UBS Securities Australia Ltd		238	(25)	Ordinary
10-301-14	UBS Securities Australia Ltd	Sell	15,257	(1,606)	Ordinary
7 - 14	UBS Securities Australia Ltd	Sell	11,391	(1,199)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	les.	005'6	(1,000)	Ordinary
0-101-14	UBS Securities Australia Ltd	Sell	5,482	(577)	Ordinary
0-701-14	UBS Securities Australia Ltd	Sell	1,302	(137)	Ordinary
0-101-14	UBS Securities Australia Ltd	Sell	3,230	(340)	Ordinary
10-701-14 0 11: 4:	UBS Securities Australia Ltd	Sell	12,996	(1,368)	Ordinary
41-14	UBS Securities Australia Ltd	Sell	8,959	(943)	Ordinary
1-10-0	UBS Securities Australia Ltd	Ę.	12,464	(1,312)	Ordinary
-in-in-	los securités Australia Ltd	Jac.	3,164	(333)	Ordinary
1 1 1	Cos securities Australia Ltd	Sell Sell	10	E	Ordinary
1 1 1	Upo Securities Australia Ltd	Sell	3,126	(329)	Ordinary
0.111	UBS Securities Australia Ltd	Sell Sell	2,945	(310)	Ordinary
1 1 1	UBS Complete Australia Ltd) Sell	1,264	(133)	Ordinary
111111111111111111111111111111111111111	LIDS Securities Australia Ltd	- Ne	3,183	(335)	Ordinary
P. I. I. I.	I IBC Committee Australia 144	Sel.	31,825	(3,350)	Ordinary
11/1/17	1 100 Counting Australia and	liac	3,183	(325)	Ordinary
1 1 1	Cos securities Australia Ltd	. Xeii	9,035	(951)	Ordinary
11-10-0	Obsideration Australia Ltd	Nell Nell	1,501	(158)	Ordinary
11-10-0	Ups Securities Australia Ltd	- T	1,416	(149)	Ordinary
1111	I IDC Constitution Australia Ltd	ner o	7,401	(6//)	Ordinary
11.10	Ups securities Australia Ltd	Jell	696	(102)	Ordinary
4 2	UBS Securities Australia Ltd	- R	770	(81)	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Sell	8,816	(928)	Ordinary
III-14	UBS Securities Australia Ltd	Sell	\$,085	(821)	Ordinary
ul-14	UBS Securities Australia Ltd	Sell	181	(61)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Setl	15,305	(1.611)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Sell	21,329	(2.244)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Sell	1,777	(187)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Sell	16.701	(1,758)	Ordinary
ul-14	UBS Securities Australia Ltd	Self	3,183	(332)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Sell	3.173	(334)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Sell	1,236	(130)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	les	3.040	(320)	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Sell	865 69	(7 305)	Ordinary
0-Jul-14	UBS Securities Australia 1td	ley.	90	(0)	Croling of
10-Jul-14	UBS Securities Australia I td	- E	342	(98)	Ordinary
0-jul-14	UBS Securities Australia I to	100	5 340	(05)	Ordinary
10-Jul-14	UBS Securities Australia Ltd	13	181	(10)	Ordinary
0-Jul-14	UBS Securities Australia Itd	3	210	(50)	Ordinary
11-14	LiBS Securities Australia Ind	8 3	613	(23)	Orginary
0.14.14	I DC Cocumition Australia Ltd		0//	(18)	Ordinary
1111	I IDC Committee Australia Ltd	===	05E,1	(140)	Ordinary
0.111	1 DC Counties Australia Ltd	Jac.	791	700	Ordinary
1	UBS Securities Australia Ltd	Jell	865	(91)	Ordinary
1	Obsideration Australia Ltd	Self	700	(23)	Ordinary
10-Jul-14	The state of the s				

1985 1985		De Ground Asser, Management UBS Wealth Management Australia Ltd BS Wealth Management Australia Ltd BS Securities Australia Ltd	Buy	1,115	117	Ordinary
USS Securities Australia Ltd Self 1,370 1,350 1,380		JBS Wealth Management Australia Ltd JBS Securities Australia Ltd	Bis	011,1	,	Olumary
USS Securities Australia Ltd Sell 13,009 (1,386)		JBS Securities Australia Ltd		1910	120	Ordinan
UBS Securities Australia Ltd Sell 25/15/25 (1.52) UBS Securities Australia Ltd Sell 25/15/25 (1.52) UBS Securities Australia Ltd Sell 133 (1.44) UBS Securities Australia Ltd Sell 134 (1.22) UBS Securities Australia Ltd Sell 139 (1.52) UBS Securities Australia Ltd Sell 1,735 (1.62) UBS Securities Australia Ltd Sell 1,735 (1.62) UBS Securities Aust			Sell	13 209	(1 386)	Ordinary
USS Securities Australia Ltd Self 143 (12) (15)		JBS Securities Australia Ltd	9	88	(D)	Ordina
UBS Securities Australia Ltd Sell 1.43 (1.2) UBS Securities Australia Ltd Sell 1.91 (1.2) UBS Securities Australia Ltd Sell 1.91 (1.2) UBS Securities Australia Ltd Sell 1.91 (1.2) UBS Securities Australia Ltd Sell 1.20 (1.5) UBS Securities Australia Ltd Sell 1.23 (1.5) UBS Securities Australia Ltd Sell 1.23 (1.5) UBS Securities Australia Ltd Sell 1.20 (1.5) UBS Securities Australia Ltd Sell 1.172 (1.5) UBS Securities Australia Ltd Sell 1.172 (1.5) UBS Securities Australia Ltd Sell 1.172 (1.5) UBS Securities Australia Ltd Sell 1.00 (1.0) UBS Securities Australia Ltd Sell 1.00 (1.0) UBS Securities Australia Ltd Sell 1.00 (1.0) UBS Securities Australia Ltd Sell 2.00 (1.0) UBS Securities Australia Ltd		IBS Securities Australia 14	19	05 30	(E) (C)	o di ilia
UBS Securities Australia Ltd Sell 114 (12) 108 Securities Australia Ltd Sell 114 (12) 108 Securities Australia Ltd Sell 116 (12) 108 Securities Australia Ltd Sell 176 (12) 108 Securities Australia Ltd Sell 1,439 (15) 108 Securities Australia Ltd Sell 1,630 (15) 108 Securities Australia Ltd Sell 1,670 (10) 109 Securities Australia Ltd Sell 1,670 (10) 108 Securities Australia Ltd Sell 1,670 (10) 109 Securities Australia Ltd Sell<		BS Securities Australia I to	logi	25,130	(2,0,2)	A CIGILIAN
UBS Securities Australia Ltd Sell 133 (14) UBS Securities Australia Ltd Sell 1191 (20) UBS Securities Australia Ltd Sell 1245 (120) UBS Securities Australia Ltd Sell 11,727 (123) UBS Securities Australia Ltd Sell 11,727 (123) UBS Securities Australia Ltd Sell 1,727 (141) UBS Securities Australia Ltd Sell 1,728 (141) UBS Securities Australia Ltd		ISC Constitute Australia Ltd	ll o	141	(01)	Ordinary
UBS Securities Australia Ltd Sell 193 (12) 108 Securities Australia Ltd Sell 114 (12) 108 Securities Australia Ltd Sell 170 (120) 108 Securities Australia Ltd Sell 173 (120) 108 Securities Australia Ltd Sell 173 (123) 108 Securities Australia Ltd Sell 173 (123) 108 Securities Australia Ltd Sell 173 (123) 109 Securities Australia Ltd Sell 173 (123) 108 Securities Australia Ltd Sell 175 (123) 108 Securities Australia Ltd Sell 1,010 (106) 108 Securities Australia Ltd Sell 3,107 (13) 108 Securities Australia Ltd Sell 3,107 (13) 108 Securities Australia Ltd Sell 3,107 (10) 108 Securities Australia Ltd Sell 3,107 (10) 108 Securities Australia Ltd Sell 3,107 (10) 109 Securities Australia Ltd Sell <td></td> <td>IOS Constituto Australia 144</td> <td>1100</td> <td>*</td> <td>(12)</td> <td>Crainary</td>		IOS Constituto Australia 144	1100	*	(12)	Crainary
UBS Securities Autoriale Ltd Sell 191 (2.0) UBS Securities Autoriale Ltd Sell 210 (12.2) UBS Securities Autoriale Ltd Sell 12.475 (1.500) UBS Securities Autoriale Ltd Sell 1.72 (1.51) UBS Securities Autoriale Ltd Sell 1.72 (1.52) UBS Securities Autoriale Ltd Sell 1.70 (1.52) UBS Securities Autoriale Ltd Sell 1.71 (1.71) UBS Securities Autoriale Ltd Sell 1.71 (1.71) UBS Securities Autoriale Ltd Sell 1.72 (1.72) UBS Securities Autoriale Ltd		DOS SECULIOS AUSTRAIA LIO	lac la	553	(14)	Ordinary
UBS Securities Australia Ltd Sell 1114 (12) UBS Securities Australia Ltd Sell 1775 (129) UBS Securities Australia Ltd Sell 1,429 (151) UBS Securities Australia Ltd Sell 1,429 (152) UBS Securities Australia Ltd Sell 1,429 (153) UBS Securities Australia Ltd Sell 1,472 (153) UBS Securities Australia Ltd Sell 1,472 (153) UBS Securities Australia Ltd Sell 1,472 (165) UBS Securities Australia Ltd Sell 1,670 (17) UBS Securities Australia Ltd Sell 1,784 (18) UBS Securities Australia Ltd Sell 1,784 (18) UBS Securities Australia Ltd		Jos Securities Australia Ltd	Sell	161	(70)	Ordinary
USS Securities Australia Ltd Sell 270 (25) USS Securities Australia Ltd Sell 7475 (1590) USS Securities Australia Ltd Sell 1,172 (15,14) (15,14) USS Securities Australia Ltd Sell 1,172 (15,14) (15,14) USS Securities Australia Ltd Sell 1,172 (15,14)		JBS Securities Australia Ltd	Sell	114	(12)	Ordinary
UBS Securities Australia Ltd Sell 175 (1520) UBS Securities Australia Ltd Sell 1,439 (1531) UBS Securities Australia Ltd Sell 1,172 (1530) UBS Securities Australia Ltd Sell 1,172 (1520) UBS Securities Australia Ltd Sell 1,172 (1521) UBS Securities Australia Ltd Sell 6,537 (667) UBS Securities Australia Ltd Sell 6,377 (670) UBS Securities Australia Ltd Sell 6,377 (670) UBS Securities Australia Ltd Sell 1,070 (10) UBS Securities Australia Ltd <td></td> <td>JBS Securities Australia Ltd</td> <td>Sell</td> <td>210</td> <td>(22)</td> <td>Ordinary</td>		JBS Securities Australia Ltd	Sell	210	(22)	Ordinary
UBS Securities Australia Ltd Sell 11,475 (1,309) UBS Securities Australia Ltd Sell 1,434 (1,534) UBS Securities Australia Ltd Sell 1,1351 (1,534) UBS Securities Australia Ltd Sell 1,1351 (1,534) UBS Securities Australia Ltd Sell 1,177 (1,534) UBS Securities Australia Ltd Sell 1,077 (1,011) UBS Securities Australia Ltd Sell 3,107 (1,011) UBS Securities Australia Ltd Sell 1,010 (1,011) UBS Securities Australia Ltd Sell 1,025 (1,011) UBS Securities Australia Ltd Sell 1,026 (1,011) UBS Securities Australia Ltd Sell 1,026 (1,011) UBS Securities Australia Ltd Sell 1,026 (1,011)		JBS Securities Australia Ltd	Sell	9/	(8)	Ordinary
UBS Securities Australia Ltd Sell 1,439 (151) DBS Securities Australia Ltd Sell 1,172 (123) UBS Securities Australia Ltd Sell 1,172 (123) UBS Securities Australia Ltd Sell 6,357 (627) UBS Securities Australia Ltd Sell 6,357 (627) UBS Securities Australia Ltd Sell 6,377 (627) UBS Securities Australia Ltd Sell 7,070 (728) UBS Securities Australia Ltd Sell 1,070 (700) UBS Securities Australia Ltd		JBS Securities Australia Ltd	Self	12 475	(1 309)	Ordinary
USS Securities Australia Ltd Sell 1,372 (1.254) USS Securities Australia Ltd Sell 1,1951 (1.254) USS Securities Australia Ltd Sell 6,357 (667) USS Securities Australia Ltd Sell 6,357 (667) USS Securities Australia Ltd Sell 3,107 (1.266) USS Securities Australia Ltd Sell 3,107 (1.076) USS Securities Australia Ltd Sell 3,07 (1.071) USS Securities Australia Ltd Sell 3,07 (1.420) USS Securities Australia Ltd Sell 3,07 (1.421) USS Securities Australia Ltd Sell 2,478 (1.421) USS Securities Australia Ltd Sell 2,132 (1.421) USS	П	IBS Securities Australia I to	Coll	OCK I	(1004)	(Similar)
UBS Securities Australia Ltd Sell 11,354 (1,254) UBS Securities Australia Ltd Sell 1,172 (1,254) UBS Securities Australia Ltd Sell 6,357 (667) UBS Securities Australia Ltd Sell 6,357 (667) UBS Securities Australia Ltd Sell 7,010 (108) UBS Securities Australia Ltd Sell 7,010 (101) UBS Securities Australia Ltd Sell 7,010 (101) UBS Securities Australia Ltd Sell 1,010 (101) UBS Securities Australia Ltd Sell 2,078 (101) UBS Securities Australia Ltd Sell 3,107 (326) UBS Securities Australia Ltd Sell 2,078 (1,011) UBS Securities Australia Ltd Sell 1,353 (1,42) UBS Securities Australia Ltd Sell 1,353 (1,42) UBS Securities Australia Ltd Sell 1,353 (1,42) UBS Securities Australia Ltd Sell 1,353 (1,60) UBS Securities Au	T	OCCUPATION AND PROPERTY AND PRO	150	654,1	(101)	Cruraly
UBS Securities Australia Ltd Sell 1,1551 (1,254) UBS Securities Australia Ltd Sell 6,357 (657) UBS Securities Australia Ltd Sell 6,357 (670) UBS Securities Australia Ltd Sell 6,357 (670) UBS Securities Australia Ltd Sell 75 (670) UBS Securities Australia Ltd Sell 100 (101) UBS Securities Australia Ltd Sell 100 (101) UBS Securities Australia Ltd Sell 2,478 (250) UBS Securities Australia Ltd Sell 1,010 (101) UBS Securities Australia Ltd Sell 1,010 (100) UBS Securities Australia Ltd Sell 1,000 (1,010) UBS Securities Australia Ltd Sell 1,000 (1,000) UBS Securities Austral		ADS SECULIFIES AUSINAIIA LIO	Sell	334	(35)	Ordinary
UBS Securities Australia Ltd Sell 1,172 (627) UBS Securities Australia Ltd Sell 6,357 (627) UBS Securities Australia Ltd Sell 3,107 (329) UBS Securities Australia Ltd Sell 70 (70 UBS Securities Australia Ltd Sell 70 (70 UBS Securities Australia Ltd Sell 2,98 (10,11) UBS Securities Australia Ltd Sell 3,107 (700) UBS Securities Australia Ltd Sell 3,107 (700) UBS Securities Australia Ltd Sell 1,010 (1,011) UBS Securities Australia Ltd Sell 1,020 (1,011) UBS Securities Australia Ltd Sell 1,020 (1,011) UBS Securities Australia Ltd Sell 1,020 (1,011) UBS Securities Australia Ltd Sell 1,135 (1,42) UBS Securities Australia Ltd Sell 1,136 (1,011) UBS Securities Australia Ltd Sell 1,136 (1,011) UBS Securities Australia	1	JBS Securities Australia Ltd	Sell	11,951	(1,254)	Ordinary
UBS Securities Australia Ltd Sell 6.357 (667) UBS Securities Australia Ltd Sell 3,107 (326) UBS Securities Australia Ltd Sell 76 (80) UBS Securities Australia Ltd Sell 667 (70) UBS Securities Australia Ltd Sell 1,010 (11) UBS Securities Australia Ltd Sell 1,010 (11) UBS Securities Australia Ltd Sell 1,010 (100) UBS Securities Australia Ltd Sell 1,010 (100) UBS Securities Australia Ltd Sell 1,010 (100) UBS Securities Australia Ltd Sell 1,020 (100) UBS Securities Australia Ltd Sell 1,020 (100) UBS Securities Australia Ltd Sell 1,020 (100) UBS Securities Australia Ltd Sell 1,010 (142) UBS Securities Australia Ltd Sell 1,020 (142) UBS Securities Australia Ltd Sell 1,010 (142) UBS Securities Australia Ltd	1	JBS Securities Australia Ltd	Sell	1,172	(123)	Ordinary
UBS Securities Australia Ltd Sell 6.357 (667) UBS Securities Australia Ltd Sell 7.07 (356) UBS Securities Australia Ltd Sell 7.07 (10) UBS Securities Australia Ltd Sell 2.07 (10) UBS Securities Australia Ltd Sell 3.07 (41) UBS Securities Australia Ltd Sell 3.07 (350) UBS Securities Australia Ltd Sell 3.07 (320) UBS Securities Australia Ltd Sell 1.95 (1,011) UBS Securities Australia Ltd Sell 1.95 (1,011) UBS Securities Australia Ltd Sell 1.95 (1,011) UBS Securities Australia Ltd Sell 2.423 (1,62) UBS Securities Australia Ltd Sell 1.95 (1,011) UBS Securities Australia Ltd		JBS Securities Australia Ltd	Sell	756'9	(99)	Ordinary
UBS Securities Australia Ltd Sell 3,107 (326) UBS Securities Australia Ltd Sell 76 (8) UBS Securities Australia Ltd Sell 1,07 (10) UBS Securities Australia Ltd Sell 1,07 (11) UBS Securities Australia Ltd Sell 1,07 (10) UBS Securities Australia Ltd Sell 1,07 (10) UBS Securities Australia Ltd Sell 1,07 (10) UBS Securities Australia Ltd Sell 1,300 (2,000) UBS Securities Australia Ltd Sell 1,382 (1,41) UBS Securities Australia Ltd Sell 1,382 (1,42) UBS Securities Australia Ltd Sell 1,382 (1,42) UBS Securities Australia Ltd Sell 1,580 (1,42) UBS Securities Australia Ltd Sell 1,580 (1,42) UBS Securities Australia Ltd Sell 3,60 (1,60) UBS Securities Australia Ltd Sell 3,60 (1,60) UBS Securities Australia Ltd		JBS Securities Australia Ltd	Sell	6,357	(99)	Ordinary
UBS Securities Australia Ltd Sell 76 (8) UBS Securities Australia Ltd Sell 1,010 (10) UBS Securities Australia Ltd Sell 1,010 (10) UBS Securities Australia Ltd Sell 9,635 (1,011) UBS Securities Australia Ltd Sell 3,107 (3,00) UBS Securities Australia Ltd Sell 1,900 (2,000) UBS Securities Australia Ltd Sell 1,363 (1,011) UBS Securities Australia Ltd Sell 1,363 (1,010) UBS Securities Australia Ltd Sell 1,754 (1,500) UBS Securities Australia Ltd Sell 1,754 (1,500) UBS Securities Australia Ltd Sell 2,742 (1,500) UBS Securities Australia Ltd Sell 2,742 (1,500) UBS Securities	ı	JBS Securities Australia Ltd	Sell	3.107	(326)	Ordinary
UBS Securities Australia Ltd Sell 667 (70) UBS Securities Australia Ltd Sell 1,010 (110) UBS Securities Australia Ltd Sell 1,010 (110) UBS Securities Australia Ltd Sell 2,78 (1,011) UBS Securities Australia Ltd Sell 9,635 (1,011) UBS Securities Australia Ltd Sell 9,635 (1,011) UBS Securities Australia Ltd Sell 9,635 (1,011) UBS Securities Australia Ltd Sell 1,306 (1,011) UBS Securities Australia Ltd Sell 1,352 (1,43) UBS Securities Australia Ltd Sell 1,135 (1,43) UBS Securities Australia Ltd Sell 2,772 (1,540) UBS Securities Australia Ltd Sell 2,772 (1,540) UBS Securitie	ı	BS Securities Australia Ltd	les.	76	(8)	Ordinan
UBS Securities Australia Ltd Sell 1,070 (10) UBS Securities Australia Ltd Sell 1,070 (11) UBS Securities Australia Ltd Sell 1,070 (10) UBS Securities Australia Ltd Sell 2,478 (200) UBS Securities Australia Ltd Sell 2,478 (200) UBS Securities Australia Ltd Sell 2,478 (200) UBS Securities Australia Ltd Sell 1,900 (2,000) UBS Securities Australia Ltd Sell 1,900 (2,000) UBS Securities Australia Ltd Sell 1,900 (1,011) UBS Securities Australia Ltd Sell 1,900 (1,000) UBS Securities Australia Ltd Sell 1,715 (1,000) UBS Securities Australia Ltd Sell 1,715 (1,000) UBS Securities Australia Ltd Sell 1,754 (1,800) UBS Securities Australia Ltd Sell 1,754 (1,500) UBS Securities Australia Ltd Sell 2,723 (1,500) UBS Securitie	П	IRS Securities Australia 14d	وا	733	0,00	Ordinary
UBS Securities Australia Ltd Sell 1,010 (105) UBS Securities Australia Ltd Sell 3,02 (11) UBS Securities Australia Ltd Sell 3,02 (10) UBS Securities Australia Ltd Sell 2,478 (260) UBS Securities Australia Ltd Sell 2,478 (260) UBS Securities Australia Ltd Sell 1,9060 (2,000) UBS Securities Australia Ltd Sell 1,9060 (2,000) UBS Securities Australia Ltd Sell 1,362 (1,011) UBS Securities Australia Ltd Sell 1,382 (1,45) UBS Securities Australia Ltd Sell 1,382 (1,45) UBS Securities Australia Ltd Sell 1,382 (1,45) UBS Securities Australia Ltd Sell 1,15 (1,45) UBS Securities Australia Ltd Sell 2,25 (1,45) UBS Securities Australia Ltd Sell 2,27 (1,55) UBS Securities Australia Ltd Sell 2,27 (1,50) UBS Securities Austra	Ť	IDC Copyrition Appropria 14d	100	000	(6)	Cidinary
UBS Securities Australia Ltd Sell 1,010 (10b) UBS Securities Australia Ltd Sell 391 (41) UBS Securities Australia Ltd Sell 2,478 (260) UBS Securities Australia Ltd Sell 2,478 (260) UBS Securities Australia Ltd Sell 3,107 (300) UBS Securities Australia Ltd Sell 3,107 (300) UBS Securities Australia Ltd Sell 1,360 (2,000) UBS Securities Australia Ltd Sell 1,365 (401) UBS Securities Australia Ltd Sell 1,365 (401) UBS Securities Australia Ltd Sell 1,365 (401) UBS Securities Australia Ltd Sell 1,715 (401) UBS Securities Australia Ltd Sell 1,714 (401) UBS Securities Australia Ltd<	Ť	Do securities Australia Ltu	# T	67	(3)	Ordinary
UBS Securities Australia Ltd Sell 391 (41) UBS Securities Australia Ltd Sell 2,478 (1,011) UBS Securities Australia Ltd Sell 2,478 (1,011) UBS Securities Australia Ltd Sell 1,9050 (2,000) UBS Securities Australia Ltd Sell 1,353 (145) UBS Securities Australia Ltd Sell 1,353 (145) UBS Securities Australia Ltd Sell 1,7154 (1,651) UBS Securities Australia Ltd Sell 1,7154 (1,552) <	ı	as securities Australia LIU	Sell	010,1	(anii)	Ordinary
UBS Securities Australia Ltd Sell 9633 (41) UBS Securities Australia Ltd Sell 2,478 (1011) UBS Securities Australia Ltd Sell 2,478 (2600) UBS Securities Australia Ltd Sell 19,060 (2,000) UBS Securities Australia Ltd Sell 19,060 (2,000) UBS Securities Australia Ltd Sell 1,353 (1,011) UBS Securities Australia Ltd Sell 1,362 (1,011) UBS Securities Australia Ltd Sell 1,382 (1,42) UBS Securities Australia Ltd Sell 1,382 (1,42) UBS Securities Australia Ltd Sell 1,553 (1,011) UBS Securities Australia Ltd Sell 1,7,154 (1,553) UBS Securities Australia Ltd Sell 3,702 (3,590)		JBS Securities Australia Ltd	Sell	105	(11)	Ordinary
UBS Securities Australia Ltd Seal 9,635 (1,011) UBS Securities Australia Ltd Seal 2,736 (2,600) UBS Securities Australia Ltd Seal 3,107 (326) UBS Securities Australia Ltd Seal 1,9060 (2,000) UBS Securities Australia Ltd Seal 1,9060 (2,000) UBS Securities Australia Ltd Seal 1,382 (1,42) UBS Securities Australia Ltd Seal 1,7154 (1,65) UBS Securities Australia Ltd Seal 2,423 (1,520) UBS Securities Australia Ltd Seal 2,423 (1,521) UBS Securities Australia Ltd Seal 2,423 (1,524) UBS Securities Australia Ltd Seal 3,143 (1,523) UBS	-1	JBS Securities Australia Ltd	Sell	391	(41)	Ordinary
UBS Securities Australia Ltd Sell 2,478 (260) UBS Securities Australia Ltd Sell 3,107 (326) UBS Securities Australia Ltd Sell 19,060 (2,000) UBS Securities Australia Ltd Sell 19,060 (2,000) UBS Securities Australia Ltd Sell 1,385 (1,011) UBS Securities Australia Ltd Sell 1,385 (1,401) UBS Securities Australia Ltd Sell 1,385 (1,401) UBS Securities Australia Ltd Sell 1,385 (1,401) UBS Securities Australia Ltd Sell 1,7154 (1,400) UBS Securities Australia Ltd Sell 2,423 (1,551) UBS Securities Australia Ltd Sell 1,7154 (1,600) UBS Securities Australia Ltd Sell 1,754 (1,600) UBS Securities Australia Ltd Sell 1,754 (1,553) UBS Securities Australia Ltd Sell 1,754 (1,553) UBS Securities Australia Ltd Sell 1,754 (1,553) <t< td=""><td></td><td>IBS Securities Australia Ltd</td><td>Sell</td><td>9,635</td><td>(1,011)</td><td>Ordinary</td></t<>		IBS Securities Australia Ltd	Sell	9,635	(1,011)	Ordinary
UBS Securities Australia Ltd Seil 3,107 (326) UBS Securities Australia Ltd Sell 19,060 (2,000) UBS Securities Australia Ltd Sell 19,060 (2,000) UBS Securities Australia Ltd Sell 1,355 (1,011) UBS Securities Australia Ltd Sell 1,362 (1,010) UBS Securities Australia Ltd Sell 2,859 (300) UBS Securities Australia Ltd Sell 1,362 (1,011) UBS Securities Australia Ltd Sell 1,7154 (1,000) UBS Securities Australia Ltd Sell 1,7154 (1,550) UBS Securities Australia Ltd Sell 1,4754 (1,550) UBS Securities Australia Ltd Sell 1,4754 (1,550) UBS Securities Australia Ltd Sell 1,4754 (1,550) UBS Securities Australia Ltd Sell 1,500 (1,550) UBS Securities Australia Ltd Sell 3,100 (1,550) UBS Securities Australia Ltd Sell 3,100 (1,550)		JBS Securities Australia Ltd	Sell	2,478	(360)	Ordinary
UBS Securities Australia Ltd Sell 19,060 (3200) UBS Securities Australia Ltd Sell 19,060 (2,000) UBS Securities Australia Ltd Sell 19,060 (2,000) UBS Securities Australia Ltd Sell 1,353 (142) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 1,7154 (140) UBS Securities Australia Ltd Sell 2,423 (7,571) UBS Securities Australia Ltd Sell 2,423 (7,571) UBS Securities Australia Ltd Sell 2,423 (7,571) UBS Securities Australia Ltd Sell 1,4,534 (1,553) UBS Securities Australia Ltd Sell 1,7,54 (7,571) UBS Securities Australia Ltd Sell 3,70 (5,59) UBS Securities Australia Ltd Sell 3,70 (5,69) UBS Se	Г	JBS Securities Australia Ltd	Sell	52	8	Ordinary
UBS Securities Australia Ltd Sell 19,660 (2,000) UBS Securities Australia Ltd Sell 19,680 (2,000) UBS Securities Australia Ltd Sell 1,353 (142) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 1,383 (144) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 1,7,154 (1,800) UBS Securities Australia Ltd Sell 1,7,154 (1,530) UBS Securities Australia Ltd Sell 2,423 (2,551) UBS Securities Australia Ltd Sell 1,7,154 (1,530) UBS Securities Australia Ltd Sell 1,650 (1,590) UBS Securities Australia Ltd Sell 1,670 (1,553) UBS Securities Australia Ltd Sell 3,700 (3,590) UBS Securities Australia Ltd Sell 3,420 (3,590)	Γ	JBS Securities Australia I td	lay	3 107	(3/5/	Ordinan
UBS Securities Australia Ltd Sell 9,635 (1,010) UBS Securities Australia Ltd Sell 4,098 (1,010) UBS Securities Australia Ltd Sell 1,382 (1,42) UBS Securities Australia Ltd Sell 1,382 (1,42) UBS Securities Australia Ltd Sell 3,859 (300) UBS Securities Australia Ltd Sell 1,7154 (1,800) UBS Securities Australia Ltd Sell 2,723 (101) UBS Securities Australia Ltd Sell 2,733 (1,501) UBS Securities Australia Ltd Sell 2,733 (1,501) UBS Securities Australia Ltd Sell 2,734 (1,501) UBS Securities Australia Ltd Sell 2,734 (1,501) UBS Securities Australia Ltd Sell 2,734 (1,501) UBS Securities Australia Ltd Sell 3,720 (5,531) UBS Securities Australia Ltd Sell 3,720 (5,531) UBS Securities Australia Ltd Sell 3,720 (350) UBS S	Γ	IBS Securities Australia I td	Coll	10,060	(000 0)	Contrary.
UBS Securities Australia Ltd Sell 19,050 (1,011) UBS Securities Australia Ltd Sell 1,353 (142) UBS Securities Australia Ltd Sell 1,353 (143) UBS Securities Australia Ltd Sell 1,385 (140) UBS Securities Australia Ltd Sell 1,382 (300) UBS Securities Australia Ltd Sell 1,7154 (1,800) UBS Securities Australia Ltd Sell 2,423 (7,571) UBS Securities Australia Ltd Sell 2,423 (7,571) UBS Securities Australia Ltd Sell 1,4,534 (1,550) UBS Securities Australia Ltd Sell 1,4,754 (1,553) UBS Securities Australia Ltd Sell 1,4,754 (1,553) UBS Securities Australia Ltd Sell 1,754 (1,553) UBS Securities Australia Ltd Sell 1,754 (1,553) UBS Securities Australia Ltd Sell 1,754 (1,553) UBS Securities Australia Ltd Sell 3,164 (350)	ı	DC Countries Australia 14d	100	2000	(2,000)	Geliaiy
UBS Securities Australia Ltd Sell 19,060 (1,420) UBS Securities Australia Ltd Sell 1,382 (1,42) UBS Securities Australia Ltd Sell 2,859 (300) UBS Securities Australia Ltd Sell 2,859 (300) UBS Securities Australia Ltd Sell 72,152 (1,571) UBS Securities Australia Ltd Sell 72,152 (7,571) UBS Securities Australia Ltd Sell 72,152 (7,571) UBS Securities Australia Ltd Sell 74,33 (7,571) UBS Securities Australia Ltd Sell 14,754 (1,580) UBS Securities Australia Ltd Sell 14,754 (1,553) UBS Securities Australia Ltd Sell 14,754 (1,553) UBS Securities Australia Ltd Sell 1,690 (1,061) UBS Securities Australia Ltd Sell 5,507 (7,533) UBS Securities Australia Ltd Sell 3,100 (1,561) UBS Securities Australia Ltd Sell 3,120 (3,590)	ı	DO SECULIOS Australia Eta	Jeil Jeil	CCO/6	(110/1)	Ordinary
UBS Securities Australia Ltd Sell 1,382 (430) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 1,382 (143) UBS Securities Australia Ltd Sell 1,135 (142) UBS Securities Australia Ltd Sell 1,1,154 (1800) UBS Securities Australia Ltd Sell 2,423 (1,531) UBS Securities Australia Ltd Sell 2,423 (1,531) UBS Securities Australia Ltd Sell 1,4,530 (1,550) UBS Securities Australia Ltd Sell 1,4,754 (1,553) UBS Securities Australia Ltd Sell 1,7,540 (1,553) UBS Securities Australia Ltd Sell 3,126 (350) UBS Securities Australia Ltd Sell 3,124 (350) UBS Securities Australia Ltd Sell 3,420 (350) UBS Securities Australia Ltd Sell 3,420 (360) UBS Secur	ı	IBS Securities Australia Ltd	Seil	090/61	(2,000)	Ordinary
UBS Securities Australia Ltd Sell 1,353 (142) UBS Securities Australia Ltd Sell 1,382 (142) UBS Securities Australia Ltd Sell 2,859 (300) UBS Securities Australia Ltd Sell 10,7154 (1,800) UBS Securities Australia Ltd Sell 17,154 (1,800) UBS Securities Australia Ltd Sell 2,413 (2,521) UBS Securities Australia Ltd Sell 2,423 (2,521) UBS Securities Australia Ltd Sell 1,4,534 (1,533) UBS Securities Australia Ltd Sell 1,4,534 (1,533) UBS Securities Australia Ltd Sell 1,4,534 (1,533) UBS Securities Australia Ltd Sell 1,5,701 (1,533) UBS Securities Australia Ltd Sell 1,5,701 (1,533) UBS Securities Australia Ltd Sell 5,501 (1,533) UBS Securities Australia Ltd Sell 5,501 (1,533) UBS Securities Australia Ltd Sell 3,154 (350)	1	IBS Securities Australia Ltd	Sell	4,098	(430)	Ordinary
UBS Securities Australia Ltd Seil 1,382 (145) UBS Securities Australia Ltd Sell 2,859 (300) UBS Securities Australia Ltd Sell 105 (11) UBS Securities Australia Ltd Sell 72,152 (157) UBS Securities Australia Ltd Sell 2,1413 (1,524) UBS Securities Australia Ltd Sell 2,423 (1,523) UBS Securities Australia Ltd Sell 1,4,530 (1,530) UBS Securities Australia Ltd Sell 2,2,772 (2,534) UBS Securities Australia Ltd Sell 22,772 (2,537) UBS Securities Australia Ltd Sell 22,772 (2,587) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securit	1	IBS Securities Australia Ltd	Sell	1,353	(142)	Ordinary
UBS Securities Australia Ltd Sell 2.859 (30) UBS Securities Australia Ltd Sell 1.05 (10) UBS Securities Australia Ltd Sell 1.154 (1,800) UBS Securities Australia Ltd Sell 2.423 (1,501) UBS Securities Australia Ltd Sell 2.423 (1,501) UBS Securities Australia Ltd Sell 1.4,534 (1,530) UBS Securities Australia Ltd Sell 1.4,534 (1,530) UBS Securities Australia Ltd Sell 1.4,754 (1,530) UBS Securities Australia Ltd Sell 2.2,23 (2,548) UBS Securities Australia Ltd Sell 1.6,500 (1,553) UBS Securities Australia Ltd Sell 3.164 (350) UBS Securities Australia Ltd Sell 3.164 (350) UBS Securities Australia Ltd Sell 3.154 (350) UBS Securities Australia Ltd Sell 3.154 (350) UBS Securities Australia Ltd Sell 3.154 (350) UBS Se	7	IBS Securities Australia Ltd	Seli	1,382	(145)	Ordinary
UBS Securities Australia Ltd Sell 105 (4) UBS Securities Australia Ltd Sell 17,154 (1800) UBS Securities Australia Ltd Sell 17,154 (1800) UBS Securities Australia Ltd Sell 2,423 (7,571) UBS Securities Australia Ltd Sell 14,73 (7,571) UBS Securities Australia Ltd Sell 14,73 (7,534) UBS Securities Australia Ltd Sell 14,754 (1,533) UBS Securities Australia Ltd Sell 14,754 (1,533) UBS Securities Australia Ltd Sell 2,275 (7,533) UBS Securities Australia Ltd Sell 5,507 (7,533) UBS Securities Australia Ltd Sell 5,507 (7,533) UBS Securities Australia Ltd Sell 3,420 (3,503) UBS Securities Australia Ltd Sell 3,420 (3,503) UBS Securities Australia Ltd Sell 3,420 (3,503) UBS Securities Australia Ltd Sell 3,154 (3,503) UB		IBS Securities Australia Ltd	Sell	2,859	(300)	Ordinary
UBS Securities Australia Ltd Sell 17,154 (11) UBS Securities Australia Ltd Sell 72,154 (1,800) UBS Securities Australia Ltd Sell 72,153 (2,254) UBS Securities Australia Ltd Sell 2,423 (2,254) UBS Securities Australia Ltd Sell 2,423 (153) UBS Securities Australia Ltd Sell 1,754 (1,530) UBS Securities Australia Ltd Sell 22,725 (2,248) UBS Securities Australia Ltd Sell 22,725 (2,285) UBS Securities Australia Ltd Sell 22,726 (2,595) UBS Securities Australia Ltd Sell 3,164 (350) UBS Securities Australia Ltd Sell 3,420 (350) UBS Securities Australia Ltd Sell 3,420 (350) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Secur		BS Securities Australia Ltd	Sell	38	(4)	Ordinary
UBS Securities Australia Ltd Sell 17,154 (1,800) UBS Securities Australia Ltd Sell 72,152 (7,571) UBS Securities Australia Ltd Sell 2,443 (7,534) UBS Securities Australia Ltd Sell 2,423 (2,554) UBS Securities Australia Ltd Sell 14,754 (1,533) UBS Securities Australia Ltd Sell 14,754 (1,533) UBS Securities Australia Ltd Sell 12,2772 (1,533) UBS Securities Australia Ltd Sell 5,501 (7,593) UBS Securities Australia Ltd Sell 5,501 (7,593) UBS Securities Australia Ltd Sell 5,501 (7,593) UBS Securities Australia Ltd Sell 3,164 (350) UBS Securities Australia Ltd Sell 3,154 (350) UB	Ī	BS Securities Australia I Id	9	105	(1)	Ordinan
UBS Securities Australia Ltd Sell 72,152 (1,571) UBS Securities Australia Ltd Sell 2,443 (2,544) UBS Securities Australia Ltd Sell 2,423 (2,544) UBS Securities Australia Ltd Sell 14,630 (1,540) UBS Securities Australia Ltd Sell 14,630 (1,540) UBS Securities Australia Ltd Sell 23,272 (2,591) UBS Securities Australia Ltd Sell 23,772 (2,397) UBS Securities Australia Ltd Sell 3,601 (5,79) UBS Securities Australia Ltd Sell 3,420 (350) UBS Securities Australia Ltd Sell 3,420 (350) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332) UBS Secu	Ť	RS Securities Australia Itd	9	17 154	(1000)	Collinary
UBS Securities Australia Ltd Sell 27,132 (1,51) UBS Securities Australia Ltd Sell 2,423 (2,524) UBS Securities Australia Ltd Sell 1,453 (1,553) UBS Securities Australia Ltd Sell 1,4754 (1,553) UBS Securities Australia Ltd Sell 2,275 (1,553) UBS Securities Australia Ltd Sell 2,275 (2,48) UBS Securities Australia Ltd Sell 1,0,080 (1,661) UBS Securities Australia Ltd Sell 3,164 (332) UBS Securities Australia Ltd Sell 1,1,904 (1,553) UBS Securities Australia Ltd Sell 1,1,904 (1,553) UBS Securities Australia Ltd Sell 3,154 (350) UBS	Τ	DO Countition A material		101/1	(000,1)	Oldinary F
UBS Securities Australia Ltd Seli 2,413 (2,244) UBS Securities Australia Ltd Sell 2,423 (555) UBS Securities Australia Ltd Sell 1,4530 (1,540) UBS Securities Australia Ltd Sell 1,4754 (1,533) UBS Securities Australia Ltd Sell 23,256 (2,448) UBS Securities Australia Ltd Sell 5,077 (3,39) UBS Securities Australia Ltd Sell 5,077 (3,39) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities	T	IBS Securities Australia Ltd	3	72,152	(7,571)	Ordinary
UBS Securities Australia Ltd Sell 960 (101) UBS Securities Australia Ltd Sell 2,423 (551) UBS Securities Australia Ltd Sell 14,754 (1,533) UBS Securities Australia Ltd Sell 22,725 (2,488) UBS Securities Australia Ltd Sell 22,725 (2,791) UBS Securities Australia Ltd Sell 3,164 (333) UBS Securities Australia Ltd Sell 3,420 (5,285) UBS Securities Australia Ltd Sell 3,420 (1,503) UBS Securities Australia Ltd Sell 3,420 (1,604) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Secur	T	BS Securities Australia Ltd	Sell	21,413	(2,254)	Ordinary
UBS Securities Australia Ltd Sell 2,423 (255) UBS Securities Australia Ltd Sell 1,4530 (1,533) UBS Securities Australia Ltd Sell 23,256 (2,448) UBS Securities Australia Ltd Sell 22,772 (7,533) UBS Securities Australia Ltd Sell 5,501 (7,59) UBS Securities Australia Ltd Sell 3,164 (333) UBS Securities Australia Ltd Sell 3,62 (3,59) UBS Securities Australia Ltd Sell 3,164 (332) UBS Securities Australia Ltd Sell 3,420 (350) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Aus		BS Securities Australia Ltd	Sell	096	(101)	Ordinary
UBS Securities Australia Ltd Sell 14,630 (1,540) UBS Securities Australia Ltd Sell 14,754 (1,533) UBS Securities Australia Ltd Sell 23,772 (2,337) UBS Securities Australia Ltd Sell 10,080 (1,691) UBS Securities Australia Ltd Sell 3,164 (333) UBS Securities Australia Ltd Sell 3,420 (5,285) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities		BS Securities Australia Ltd	lles	2,423	(252)	Ordinary
UBS Securities Australia Ltd Seli 1,7,50 (1,533) UBS Securities Australia Ltd Seli 22,256 (2,48) UBS Securities Australia Ltd Seli 22,772 (2,937) UBS Securities Australia Ltd Seli 10,080 (1,061) UBS Securities Australia Ltd Seli 3,164 (333) UBS Securities Australia Ltd Seli 3,420 (360) UBS Securities Australia Ltd Seli 3,154 (350) UBS Securities Australia Ltd Seli 3,154 (360) UBS Securities Australia Ltd Seli 3,154 (350) UBS Securities Australia Ltd Seli 3,154 (360) UBS Securities Au	Ī	RS Securities Australia I to	ley.	14 630	(1,540)	Cardina
UBS Securities Australia Ltd Sell 1,1,5,54 UBS Securities Australia Ltd Sell 22,772 (2,397) UBS Securities Australia Ltd Sell 22,772 (2,397) UBS Securities Australia Ltd Sell 10,580 (1,599) UBS Securities Australia Ltd Sell 3,164 (333) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (350) UBS Securities Australia Ltd Sell 3,154 (350) UBS Securities Australia Ltd	T	DO Con which A retails (+)	11-1	2004	(0#0,1)	Comary
UBS Securities Australia Ltd Seli 2.2.756 (2,448) UBS Securities Australia Ltd Sell 2.2.735 (2,448) UBS Securities Australia Ltd Sell 10,080 (1,061) UBS Securities Australia Ltd Sell 3,164 (333) UBS Securities Australia Ltd Sell 11,904 (1,253) UBS Securities Australia Ltd Sell 11,904 (1,253) UBS Securities Australia Ltd Sell 3,420 (350) UBS Securities Australia Ltd Sell 3,154 (350) UBS Securit	T	bo securities Australia Ltd	Seli	14,754	(555,1)	Ordinary
UBS Securities Australia Ltd Sell 2.7.72 (2.397) UBS Securities Australia Ltd Sell 5,501 (579) UBS Securities Australia Ltd Sell 3,164 (333) UBS Securities Australia Ltd Sell 3,020 (5,285) UBS Securities Australia Ltd Sell 3,124 (360) UBS Securities Australia Ltd Sell 3,124 (360) UBS Securities Australia Ltd Sell 3,124 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Australi	1	BS Securities Australia Ltd	Self	23,256	(2,448)	Ordinary
UBS Securities Australia Ltd Sell 5.501 (579) UBS Securities Australia Ltd Sell 1,0,860 (1,661) UBS Securities Australia Ltd Sell 3,164 (333) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia	_	BS Securities Australia Ltd	Sell	27,22	(2,397)	Ordinary
UBS Securities Australia Ltd Sell 1,0,980 (1,061) UBS Securities Australia Ltd Sell 3,64 (333) UBS Securities Australia Ltd Sell 50,208 (5,285) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (320) UBS Securities Australia Ltd Sell 3,154 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Austral	_	BS Securities Australia Ltd	Sell	5,501	(625)	Ordinary
UBS Securities Australia Ltd Sell 3,164 (333) UBS Securities Australia Ltd Sell 50,208 (5,285) UBS Securities Australia Ltd Sell 3,420 (5,865) UBS Securities Australia Ltd Sell 11,504 (1,533) UBS Securities Australia Ltd Sell 3,154 (350) UBS Securities Australia Ltd Sell 3,154 (322) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Austr	Ē	BS Securities Australia Ltd	Sell	10.080	(1.061)	Ordinary
UBS Securities Australia Ltd Sell 50,008 (5,285) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 11,904 (1,533) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (322) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (322) UBS Securities Australia Ltd Sell 3,154 (350)	f	BS Sequrities Australia Ltd	las	3 164	(686)	Ording
UBS Securities Australia Ltd Sell 3,420 (3,53) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332)	Ť	BS Securities Australia I tol	Coll	50.708	(2002)	dia in
UBS Securities Australia Ltd Sell 3,420 (350) UBS Securities Australia Ltd Sell 11,504 (1,533) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,002 (316) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 627 (66) UBS Securities Australia Ltd Sell 627 (66) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (322) UBS Securities Australia Ltd Sell 3,154 (322) UBS Securities Australia Ltd Sell 3,154 (350) UBS Securities Australia Ltd Sell 3,154 (350) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332)	T	DO Secondides Australia Lico	1100	90,2,0e	(507'5)	Ordinary
UBS Securities Australia Ltd Sell 11,904 (1,253) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (315) UBS Securities Australia Ltd Sell 3,154 (316) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 627 (66) UBS Securities Australia Ltd Sell 3,154 (332)	T	bs securities Australia Ltd	Sell	3,420	(390)	Ordinary
UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (322) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,520 (360) UBS Securities Australia Ltd Sell 3,154 (322) UBS Securities Australia Ltd Sell 3,154 (332)	٦	BS Securities Australia Ltd	Sell	11,904	(1,253)	Ordinary
UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,002 (316) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,720 (360) UBS Securities Australia Ltd Sell 3,420 (360)	Ī	BS Securities Australia Ltd	Sell	3,420	(360)	Ordinary
UBS Securities Australia Ltd Sell 3,002 (316) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 650 UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332)		BS Securities Australia Ltd	Sell	3,154	(332)	Ordinary
UBS Securities Australia Ltd Self 3,154 (332) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 627 (66) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332)	F	BS Securities Australia Ltd	Sell	3.002	(316)	Ordinary
UBS Securities Australia Ltd Sell	Ī	RS Securities Australia Itcl	Call	3.154	(055)	, and o
UBS Securities Australia Ltd Sell 3,420 (380) UBS Securities Australia Ltd Sell 627 (66) UBS Securities Australia Ltd Sell 3,154 (332)	Ť	DO Chartella Australia 14d	10.0	101.0	1700)	Clumary
UBS Securities Australia Ltd Sell 6.27 (66) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332)	Ť	BS Securities Australia Ltd	llex.	3,420	(390)	Ordinary
UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Self 3,154 (332)		BS Securities Australia Ltd	Sell	627	(99)	Ordinary
UBS Securities Australia Ltd Sell 3,420 (360) UBS Securities Australia Ltd Sell 3,154 (332) UBS Securities Australia Ltd Sell 3,154 (332)		BS Securities Australia Ltd	les	3,154	(332)	Ordinary
UBS Securities Australia Ltd Sell 3.154 (332) UBS Securities Australia Ltd Self 3.154 (332)	Γ	BS Securities Australia Ltd	Sell	3,420	(360)	Ordinary
UBS Securities Australia Ltd Self 3.154 (332)	T	RS Sen wities Australia I to	Call	3 154	(332)	Ordinan
UBS Securities Australia Ltd Self (332)	T	or Carried Australia 14d	100	2,104	(2007)	Cigiral A
1200		BS Securities Australia Ltd	Sef	3,154	(332)	Ordinary

0-14-14	II IRS Securities Australia I to	2	7100	0.00	1
0-jul-14	UBS Securities Australia Itd	Riss	2 707	740	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	5.916	624	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	43.930	4.634	Ordinary
-14	UBS Securities Australia Ltd	Buv	13.196	1.392	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	3,792	400	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	7,812	824	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	1,431	151	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Виу	3,223	340	Ordinary
10-jul-14	UBS Securities Australia Ltd	Buy	967	102	Ordinary
-14	UBS Securities Australia Ltd	Buy	3,745	395	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	5,176	546	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	5,176	546	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	36,868	3,889	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	5,916	624	Ordinary
4	UBS Securities Australia Ltd	Buy	2,844	300	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	2,673	282	Ordinary
4	UBS Securities Australia Ltd	Buy	6,920	730	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	904,506	95,412	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	3,792	400	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	5,972	630	Ordinary
-14	UBS Securities Australia Ltd	Buy	12,153	1,282	Ordinary
-14	UBS Securities Australia Ltd	Buy	12,324	1,300	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	12,324	1,300	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	673	71	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	284	30	Ordinary
-14	UBS Securities Australia Ltd	Buy	3,792	400	Ordinary
-14	UBS Securities Australia Ltd	Buy	3,223	340	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	5,916	624	Ordinary
-14	UBS Securities Australia Ltd	Buv	11.585	1 222	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buv	2 038	215	Ordinary
14	UBS Securities Australia Ltd	Buv	2 408	254	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buv	787 665	83.087	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buv	68 265	7 201	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buv	44 433	4 687	Ordinary
4-	UBS Securities Australia Ltd	Buv	5 972	630	Ordinary
-14	I IBS Securities Australia 14d	i a	270,0	88	y and a
10-lul-14	I BS Securities Australia 11d	8.8	3,732	400	Ordinary
10-1-14	I IRS Securities Australia 11d	boy.	600,	1/3	Ordinary
1 2	I IRS Securities Australia 14d	Sugar.	52,033	245,5	Crainary
10-10-1	I IBC Societion Australia Ltd	Buy	7,034	742	Ordinary
<u>.</u> ;	UBS Securities Australia Ltd	Buy	5,187	546	Ordinary
10-701-14	UBS Securities Australia Ltd	Buy	19,475	2,050	Ordinary
1 :	Obsider Australia Ltd	Buy	5,985	069	Ordinary
t0-101-0	UBS Securities Australia Ltd	Sell	205,245	(21,582)	Ordinary
10-701-14	UBS Securities Australia Ltd	Sell	7,579	(797)	Ordinary
<u>-</u> :	UBS Securities Australia Ltd	Sell	95	(10)	Ordinary
10-701-14	UBS securities Australia Ltd	Self	3,804	(400)	Ordinary
<u>+</u> :	UBS Securities Australia Ltd	Sell Sell	28,330	(2,979)	Ordinary
0-301-14	UBS Securities Australia Ltd	Sell	7,950	(928)	Ordinary
10-70-14	UBS Securities Australia Ltd	Sell	5,991	(930)	Ordinary
41-In-14	UBS Securities Australia Ltd	Sell	418	(44)	Ordinary
: :	Upo Securities Australia Ltd	Neil Neil	1,845	(194)	Ordinary
41-101-0	UBS Securities Australia Ltd	Buy	2888	62	Ordinary
4	UBS Securities Australia Ltd	Buy	31,213	3,289	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	674	71	Ordinary
10-Jul- 14	UBS Securities Australia Ltd	Buy	33,158	3,494	Ordinary
11 17	ribs securities Australia Ltd	Buy	877	24	Ordinary
0-701-14	Ligg Securities Australia Lig	Buy	2,135	225	Ordinary
4 :	UBS Securities Australia Ltd	Buy	26,980	2,843	Ordinary
4-10I-0	UBS Securities Australia Ltd	Buy	1,518	160	Ordinary
10-JUI-14	UBS Securities Australia Ltd	Buy	19	2	Ordinary
1	UBS Securities Australia Ltd	Buy	5,922	624	Ordinary
- 1	UBS Securities Australia Ltd	Buy	712	75	Ordinary
11.10				`	,
	IUBS Securities Australia Ltd	Buv	σ	1	Ordinary

USS Scorrides Australia Ltd By J 2,555 USS Scorrides Australia Ltd Buy 2,550 USS Scorrides Australia Ltd Buy 1,550 1,517 USS Scorrides Australia Ltd Buy 1,520 1,517 USS Scorrides Australia Ltd Buy 1,526 1,500 USS Scorrides Australia Ltd Buy 1,526 1,100 USS Scorrides Australia Ltd Buy 1,526 1,100 USS Scorrides Australia Ltd Buy 4,520 2,500 USS Scorrides Australia Ltd Buy 4,572 2,500 USS Scorrides Australia Ltd Buy 4,572 4,572 USS Scorrides Australia Ltd Buy 4,572 4,572 USS Scorrides Australia Ltd Buy 4,572 2,500 USS Scorrides Australia Ltd Buy 2,520 2,500			N N	321 701	01000	, included
USS Scorrides Australia Ltd Buy 2,698 2,508 2,508 USS Scorrides Australia Ltd Buy 1,720 2,508 1,517 USS Scorrides Australia Ltd Buy 1,720 1,720 2,500 USS Scorrides Australia Ltd Buy 1,720 1,720 2,500 USS Scorrides Australia Ltd Buy 1,720 1,720 2,500 USS Scorrides Australia Ltd Buy 1,720 2,500 2,500 USS Scorrides Australia Ltd Buy 1,720 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500 2,500 2,500 2,500 USS Scorrides Australia Ltd Buy 2,500	4	UBS Securities Australia 11d	S. S.	26,120	20,02	Ordinary
UBS Scorrites Australia Lud Buy 2.6,569 2.5,410 UBS Scorrites Australia Lud Buy 1,250 1,517 UBS Scorrites Australia Lud Buy 1,239 1,110 UBS Scorrites Australia Lud Buy 1,123 1,110 UBS Scorrites Australia Lud Buy 1,123 1,110 UBS Scorrites Australia Lud Buy 1,123 1,110 UBS Scorrites Australia Lud Buy 4,571 2,12,24 1,110 UBS Scorrites Australia Lud Buy 4,571 3,00 4,877 4,179 UBS Scorrites Australia Lud Buy 4,571 3,00 4,877 4,179 UBS Scorrites Australia Lud Buy 4,577 4,179 4,877 4,179 UBS Scorrites Australia Lud Buy 4,587 3,00 3,00 4,00 UBS Scorrites Australia Lud Buy 2,525 2,471 3,00 4,00 UBS Scorrites Australia Lud Buy 2,350 3,00 1,00 8,00 1,00 8,00	ı	UBS Securities Australia Ltd	Buy	24,698	2,608	Ordinary
UBS Securities Australia Ltd Buy 1,356 1,517 UBS Securities Australia Ltd Buy 1,356 1,189 UBS Securities Australia Ltd Buy 1,125 1,110 UBS Securities Australia Ltd Buy 1,125 1,110 UBS Securities Australia Ltd Buy 2,132,244 3,000 UBS Securities Australia Ltd Buy 2,132,244 3,000 UBS Securities Australia Ltd Buy 2,137 4,877 UBS Securities Australia Ltd Buy 2,137 4,877 UBS Securities Australia Ltd Buy 3,578 3,001 UBS Securities Australia Ltd Buy 3,578 3,001 UBS Securities Australia Ltd Buy 3,520 3,001 UBS Securities Australia Ltd Buy 2,520 3,001 UBS Securities Australia Ltd	П	UBS Securities Australia Ltd	Buy	56,696	2,819	Ordinary
UBS Scorrities Australia Lud Buy 14,356 1,186 UBS Scorrities Australia Lud Buy 47,350 2,000 UBS Scorrities Australia Lud Buy 47,350 2,000 UBS Scorrities Australia Lud Buy 1,03,201 1,110 UBS Scorrities Australia Lud Buy 1,03,201 3,00 UBS Scorrities Australia Lud Buy 4,877 4,877 UBS Scorrities Australia Lud Buy 4,877 4,77 UBS Scorrities Australia Lud Buy 3,704 3,97 UBS Scorrities Australia Lud Buy 3,704 3,97 UBS Scorrities Australia Lud Buy 3,704 3,97 UBS Scorrities Australia Lud Buy 2,520 2,60 UBS Scorrities Australia Lud <t< td=""><td></td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>213,501</td><td>22,545</td><td>Ordinary</td></t<>		UBS Securities Australia Ltd	Buy	213,501	22,545	Ordinary
UBS Securities Australia Ltd Buy 11,250 1,188 UBS Securities Australia Ltd Buy 47,350 1,108 UBS Securities Australia Ltd Buy 10,512 1,101 UBS Securities Australia Ltd Buy 218,2284 23,050 UBS Securities Australia Ltd Buy 28,375 4,179 UBS Securities Australia Ltd Buy 28,375 4,179 UBS Securities Australia Ltd Buy 2,372 4,179 UBS Securities Australia Ltd Buy 3,704 3,20 UBS Securities Australia Ltd Buy 3,704 3,20 UBS Securities Australia Ltd Buy 2,20 3,01 UBS Securities Australia Ltd Buy 2,62 2,02 UBS Securities Australia Ltd Buy 2,62 2,02 UBS Securities Australia Ltd Bu	- 1	UBS Securities Australia Ltd	Buy	14,366	1,517	Ordinary
UBS Securities Australia Ltd Buy 47,320 5,000 UBS Securities Australia Ltd Buy 10,2910 1,110 UBS Securities Australia Ltd Buy 218,291 2,627 UBS Securities Australia Ltd Buy 218,292 2,000 UBS Securities Australia Ltd Buy 3,704 3,92 UBS Securities Australia Ltd Buy 3,62 3,00 UBS Securities Australia Ltd Buy 2,65 3,00 UBS Securities Australia Ltd Buy 2,650 3,00 UBS Securities Australia Ltd Buy 2,503 2,487 UBS Securities Australia Ltd Buy 2,593 2,487 UBS Securities Australia Ltd Buy 2,593 2,744 UBS Securities Australia Ltd	- 1	UBS Securities Australia Ltd	Buy	11,250	1,188	Ordinary
UBS Securities Australia Ltd Buy 10512 1,110 UBS Securities Australia Ltd Buy 102,126 2,125 UBS Securities Australia Ltd Buy 2,12,264 2,135 UBS Securities Australia Ltd Buy 3,177 4,179 UBS Securities Australia Ltd Buy 3,172 3,07 UBS Securities Australia Ltd Buy 3,272 3,07 UBS Securities Australia Ltd Buy 3,08 3,07 UBS Securities Australia Ltd Buy 2,520 1,016 UBS Securities Australia Ltd Buy 2,520 1,02 UBS Securities Australia Ltd <	١	UBS Securities Australia Ltd	Buy	47,350	2,000	Ordinary
UBS Securities Australia Ltd Buy 102,381 10,381 UBS Securities Australia Ltd Buy 219,281 20,200 UBS Securities Australia Ltd Buy 3,704 39,2 UBS Securities Australia Ltd Buy 3,705 1,011 UBS Securities Australia Ltd Buy 5,54 1,011 UBS Securities Australia Ltd Buy 5,54 1,011 UBS Securities Australia Ltd Buy 5,54 1,011 UBS Securities Australia Ltd Buy 2,592 1,016 UBS Securities Australia Ltd Buy 2,592 1,016 UBS Securities Australia Ltd Buy 2,592 1,016 UBS Securities Australia Ltd	- 1	UBS Securities Australia Ltd	Buy	10,512	1,110	Ordinary
UBS Securities Australia Ltd Buy 572,84 23,00 UBS Securities Australia Ltd Buy 467,77 4177 188 UBS Securities Australia Ltd Buy 5,54 4,77 188 UBS Securities Australia Ltd Buy 7,027 3,27 4,77 188 UBS Securities Australia Ltd Buy 7,027 7,27 1,011 UBS Securities Australia Ltd Buy 7,027 3,02 UBS Securities Australia Ltd Buy 7,027 3,07 UBS Securities Australia Ltd Buy 9,54 1,01 UBS Securities Australia Ltd Buy 9,54 1,01 UBS Securities Australia Ltd Buy 9,54 3,01 UBS Securities Australia Ltd Buy 2,523 2,67 UBS Securities Australia Ltd Buy 2,523 2,74 UBS Securities Australia Ltd Buy 2,552 2,67 UBS Securities Australia Ltd Buy 2,562 2,67 UBS Securities Australia Ltd Buy 2,523		UBS Securities Australia Ltd	Buy	102,910	10,867	Ordinary
UBS Securities Australia Ltd Buy 85772 4877 UBS Securities Australia Ltd Buy 3571 4877 UBS Securities Australia Ltd Buy 3,704 323 UBS Securities Australia Ltd Buy 7,021 7,43 UBS Securities Australia Ltd Buy 9,54 1,011 UBS Securities Australia Ltd Buy 2,895 3,00 UBS Securities Australia Ltd Buy	-14	UBS Securities Australia Ltd	Buy	218,284	23,050	Ordinary
UBS Securities Australia Lud Buy 99.575 4,776 UBS Securities Australia Lud Buy 9,574 7,971 7,83 UBS Securities Australia Lud Buy 7,071 7,43 7,61 10.1 UBS Securities Australia Lud Buy 7,237 7,43 7,61 10.1 <	-14	UBS Securities Australia Ltd	Buy	45,712	4,827	Ordinary
UBS Securities Australia Ltd Buy 9,157 967 UBS Securities Australia Ltd Buy 3,704 1,92 UBS Securities Australia Ltd Buy 7,021 1,777 1,23 UBS Securities Australia Ltd Buy 9,52 1,01 1 UBS Securities Australia Ltd Buy 2,920 1,05 UBS Securities Australia Ltd Buy 2,927 1,01 UBS Securities Australia Ltd Buy 2,625 2,83 UBS Securities Australia Ltd Buy 2,625 2,88 UBS Securities Australia Ltd Buy 2,625 2,89 UBS Securities Australia Ltd Buy 2,625 2,80 UBS Securities Australia Ltd Buy 2,625 2,74 UBS Securities Aus	-14	UBS Securities Australia Ltd	Buy	39,575	4,179	Ordinary
UBS Securities Australia Ltd Buy 3,704 379 UBS Securities Australia Ltd Buy 1,777 1,43 UBS Securities Australia Ltd Buy 7,031 1,43 UBS Securities Australia Ltd Buy 9,554 1,011 UBS Securities Australia Ltd Buy 2,927 309 UBS Securities Australia Ltd Buy 2,927 309 UBS Securities Australia Ltd Buy 2,927 309 UBS Securities Australia Ltd Buy 2,927 2,827 UBS Securities Australia Ltd Buy 2,927 2,827 UBS Securities Australia Ltd Buy 2,527 2,887 UBS Securities Australia Ltd Buy 2,527 2,887 UBS Securities Australia Ltd Buy 2,527 2,887 UBS Securities Australia Ltd Buy 2,526 1,774 UBS Securities Australia Ltd Buy 2,527 2,487 UBS Securities Australia Ltd Buy 2,527 2,587 UBS Securities Australia Ltd Buy <td>-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>9,157</td> <td>296</td> <td>Ordinary</td>	-14	UBS Securities Australia Ltd	Buy	9,157	296	Ordinary
UBS Securities Australia Ltd Buy 1,777 18B UBS Securities Australia Ltd Buy 9,524 1,011 UBS Securities Australia Ltd Buy 9,524 1,011 UBS Securities Australia Ltd Buy 2,900 1,005 UBS Securities Australia Ltd Buy 2,900 1,005 UBS Securities Australia Ltd Buy 2,900 1,005 UBS Securities Australia Ltd Buy 2,800 1,005 UBS Securities Australia Ltd Buy 2,800 1,005 UBS Securities Australia Ltd Buy 2,800 1,005 UBS Securities Australia Ltd Buy 2,650 1,005 UBS Securities Australia Ltd Buy 2,800 1,005 UBS Securities Australia Ltd <td< td=""><td>H</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,704</td><td>392</td><td>Ordinary</td></td<>	H	UBS Securities Australia Ltd	Buy	3,704	392	Ordinary
UBS Securities Australia Ltd Buy 7,221 7.43 UBS Securities Australia Ltd Buy 9,554 1,011 UBS Securities Australia Ltd Buy 2,920 309 UBS Securities Australia Ltd Buy 2,920 309 UBS Securities Australia Ltd Buy 2,532 360 UBS Securities Australia Ltd Buy 2,734 841 UBS Securities Australia Ltd Buy 2,735 2,88 5,9 UBS Securities Australia Ltd Buy 2,735 2,88 1,977 UBS Securities Australia Ltd Buy 2,655 7,74 UBS Securities Australia Ltd Buy 1,675 1,975 UBS Securities Australia Ltd Buy 1,683 1,977 UBS Securities Australia Ltd Buy 1,683 1,977 UBS Securities Australia Ltd Buy 1,683 1,977 UBS Securities Australia Ltd Buy 2,547 2,665 1,978 UBS Securities Australia Ltd Buy 2,565 2,744 1,878 </td <td>ı</td> <td>UBS Securities Australia Ltd</td> <td>Bov</td> <td>1 777</td> <td>188</td> <td>Ordinary</td>	ı	UBS Securities Australia Ltd	Bov	1 777	188	Ordinary
UBS Securities Australia Ltd Buy 9.554 1,011 UBS Securities Australia Ltd Buy 9.92 105 UBS Securities Australia Ltd Buy 2,920 105 UBS Securities Australia Ltd Buy 2,922 105 UBS Securities Australia Ltd Buy 2,722 2,88 UBS Securities Australia Ltd Buy 2,722 2,88 UBS Securities Australia Ltd Buy 2,675 1,77 UBS Securities Australia Ltd Buy 2,675 1,77 UBS Securities Australia Ltd Buy 2,675 1,77 UBS Securities Australia Ltd Buy 1,890 1,77 UBS Securities Australia Ltd Buy 1,655 1,74 UBS Securities Australia Ltd Buy 2,675 1,74 UBS Securities Australia Ltd Buy 2,78 1,74 UBS Securities Australia Ltd Buy 2,78 2,60 UBS Securities Australia Ltd Buy 2,78 2,60 UBS Securities Australia Ltd Buy <	ı	UBS Securities Australia Ltd	Buv	7 021	743	Ordinary
UBS Securities Australia Ltd Buy 992 105 UBS Securities Australia Ltd Buy 5,920 309 UBS Securities Australia Ltd Buy 1,947 841 UBS Securities Australia Ltd Buy 2,722 268 UBS Securities Australia Ltd Buy 2,722 268 UBS Securities Australia Ltd Buy 2,722 288 UBS Securities Australia Ltd Buy 2,722 2,837 UBS Securities Australia Ltd Buy 2,852 2,837 UBS Securities Australia Ltd Buy 2,855 1,977 UBS Securities Australia Ltd Buy 2,655 1,978 UBS Securities Australia Ltd Buy 2,695 1,978 UBS Securities Australia Ltd Buy 2,787 2,895 UBS Securities Australia Ltd Buy	L	UBS Securities Australia I td	Buy	13C/	101	Ordinary
UBS Securities Australia Ltd Buy 2,322 100 UBS Securities Australia Ltd Buy 2,322 100 UBS Securities Australia Ltd Buy 7,947 20 UBS Securities Australia Ltd Buy 5,88 59 UBS Securities Australia Ltd Buy 5,722 2,88 UBS Securities Australia Ltd Buy 5,650 7,47 UBS Securities Australia Ltd Buy 5,650 7,47 UBS Securities Australia Ltd Buy 5,650 7,47 UBS Securities Australia Ltd Buy 6,682 7,73 UBS Securities Australia Ltd Buy 6,682 7,74 UBS Securities Australia Ltd Buy 2,593 1,94 UBS Securities Australia Ltd Buy 5,683 1,97 UBS Securities Australia Ltd Buy <t< td=""><td>П</td><td>II IRS Societies Australia 14d</td><td>P. O.</td><td>1000</td><td>20,</td><td>Cidinaly</td></t<>	П	II IRS Societies Australia 14d	P. O.	1000	20,	Cidinaly
UBS Securities Australia Ltd Buy 1,447 206 UBS Securities Australia Ltd Buy 5,947 841 UBS Securities Australia Ltd Buy 5,947 841 UBS Securities Australia Ltd Buy 2,722 288 UBS Securities Australia Ltd Buy 2,722 288 UBS Securities Australia Ltd Buy 2,752 2,487 UBS Securities Australia Ltd Buy 2,752 2,887 UBS Securities Australia Ltd Buy 2,853 1,977 UBS Securities Australia Ltd Buy 2,653 1,978 UBS Securities Australia Ltd Buy 2,693 1,978 UBS Securities Australia Ltd Buy 2,693 1,978 UBS Securities Australia Ltd Buy 2,693 1,978 UBS Securities Australia Ltd Buy 2,797 2,695 UBS Securities Australia Ltd Buy 2,797 2,695 UBS Securities Australia Ltd Buy 2,797 2,695 UBS Securities Australia Ltd Buy <td>14</td> <td>UBS Securities Australia 1td</td> <td>W. W.</td> <td>2000</td> <td>5 5</td> <td>Ordinary</td>	14	UBS Securities Australia 1td	W. W.	2000	5 5	Ordinary
UBS Securities Australia Ltd Buy 7,347 8.47 UBS Securities Australia Ltd Buy 558 59 UBS Securities Australia Ltd Buy 2,722 288 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 2,655 281 UBS Securities Australia Ltd Buy 1,267 756 UBS Securities Australia Ltd Buy 1,830 2,613 UBS Securities Australia Ltd Buy 1,830 2,614 UBS Securities Australia Ltd Buy 1,830 2,614 UBS Securities Australia Ltd Buy 1,830 1,065 UBS Securities Australia Ltd Buy 1,830 1,065 UBS Securities Australia Ltd Buy 5,831 1,776 UBS Securities Australia Ltd Buy 5,832 1,947 UBS Securities Australia Ltd Buy	-14	UBS Securities Australia 11d	Birk	1 047	900	900
USS Securities Australia Ltd Buy 5,732 98 USS Securities Australia Ltd Buy 5,732 288 1,937 USS Securities Australia Ltd Buy 2,835 1,016 1,016 USS Securities Australia Ltd Buy 2,835 1,977 1,016 USS Securities Australia Ltd Buy 1,465 1,577 1,016 USS Securities Australia Ltd Buy 1,665 1,977 1,016 USS Securities Australia Ltd Buy 1,665 1,978 1,016 USS Securities Australia Ltd Buy 1,665 1,978 1,016 USS Securities Australia Ltd Buy 1,665 1,744 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 1,014 1,006 <td>14</td> <td>1 IRC Countries Australia 14d</td> <td>Paris,</td> <td>1,947</td> <td>700</td> <td>Ordinary</td>	14	1 IRC Countries Australia 14d	Paris,	1,947	700	Ordinary
USS Securities Australia Ltd Buy 2,732 2.88 USS Securities Australia Ltd Buy 2,732 2.88 UBS Securities Australia Ltd Buy 2,737 2,487 UBS Securities Australia Ltd Buy 2,150 2,487 UBS Securities Australia Ltd Buy 1,267 7,787 7,787 UBS Securities Australia Ltd Buy 1,689 1,977 UBS Securities Australia Ltd Buy 1,689 1,774 UBS Securities Australia Ltd Buy 2,593 1,776 UBS Securities Australia Ltd Buy 2,789 6,007 UBS Securities Australia Ltd Buy 2,789 6,007 UBS Securities Australia Ltd Buy 2,887 1,776 UBS Securities Australia Ltd Buy 2,887 1,787 2,000 <th< td=""><td></td><td>1 BC Countries Australia 14d</td><td>Pr. in</td><td>(10,7)</td><td>4 5</td><td>Ordinary</td></th<>		1 BC Countries Australia 14d	Pr. in	(10,7)	4 5	Ordinary
UBS Securities Australia Ltd Buy 5,722 2,88 UBS Securities Australia Ltd Buy 5,601 1,016 UBS Securities Australia Ltd Buy 5,605 1,577 UBS Securities Australia Ltd Buy 6,965 7,37 UBS Securities Australia Ltd Buy 1,630 2,467 UBS Securities Australia Ltd Buy 7,267 7,66 UBS Securities Australia Ltd Buy 7,267 7,66 UBS Securities Australia Ltd Buy 7,68 7,44 UBS Securities Australia Ltd Buy 7,68 7,44 UBS Securities Australia Ltd Buy 7,68 7,44 UBS Securities Australia Ltd Buy 7,78 1,78 UBS Securities Australia Ltd Buy 7,78 1,68 UBS Securities Australia Ltd Buy		Libs Societies Australia Lid	an's	000	60	Ordinary
UBS Securities Australia Ltd Buy 9,601 1,016 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 1,655 759 UBS Securities Australia Ltd Buy 1,657 769 UBS Securities Australia Ltd Buy 1,657 769 UBS Securities Australia Ltd Buy 1,657 769 UBS Securities Australia Ltd Buy 1,657 744 UBS Securities Australia Ltd Buy 1,655 1,748 UBS Securities Australia Ltd Buy 1,655 1,748 UBS Securities Australia Ltd Buy 1,665 1,748 UBS Securities Australia Ltd Buy 5,683 1,065 UBS Securities Australia Ltd Buy	<u>.</u> .	Obs Securities Australia Ltd	Buy	7,777	288	Ordinary
UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 22,502 2,487 UBS Securities Australia Ltd Buy 1,465 1,577 UBS Securities Australia Ltd Buy 1,267 7,667 7,687 <	<u>+</u> ;	UBS Securities Australia Ltd	Buy	9,601	1,016	Ordinary
UBS Securities Australia Ltd Bby 2.5502 2.487 UBS Securities Australia Ltd Buy 6.965 1.977 UBS Securities Australia Ltd Buy 1.6283 1.977 UBS Securities Australia Ltd Buy 7.267 7.69 UBS Securities Australia Ltd Buy 7.267 7.69 UBS Securities Australia Ltd Buy 1.899 2.744 UBS Securities Australia Ltd Buy 1.899 2.744 UBS Securities Australia Ltd Buy 1.899 1.006 UBS Securities Australia Ltd Buy 1.899 1.006 UBS Securities Australia Ltd Buy 1.868 1.006 UBS Securities Australia Ltd Buy 5.893 1.006 UBS Securities Australia Ltd Buy 5.893 1.006 UBS Securities Australia Ltd Buy 5.895 1.006 UBS Securities Australia Ltd Buy 5.895 1.006 UBS Securities Australia Ltd Buy 5.816 1.006 UBS Securities Australia Ltd <	-14	UBS Securities Australia Ltd	Buy	2,835	300	Ordinary
UBS Securities Australia Ltd Buy 6,965 737 UBS Securities Australia Ltd Buy 1,465 1,55 UBS Securities Australia Ltd Buy 7,267 1,697 UBS Securities Australia Ltd Buy 2,653 200 UBS Securities Australia Ltd Buy 2,693 1,978 UBS Securities Australia Ltd Buy 2,693 1,978 UBS Securities Australia Ltd Buy 2,694 1,676 UBS Securities Australia Ltd Buy 2,694 1,076 UBS Securities Australia Ltd Buy 2,694 1,076 UBS Securities Australia Ltd Buy 5,804 1,076 UBS Securities Australia Ltd Buy 5,805 1,07 UBS Securities Australia Ltd Buy 1,445 1,53 UBS Securities Australia Ltd Buy 1,445 1,53 UBS Securities Australia Ltd Buy 1,784 1,78 UBS Securities Australia Ltd Buy 2,381 1,78 UBS Securities Australia Ltd Buy <td>-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>23,502</td> <td>2,487</td> <td>Ordinary</td>	-14	UBS Securities Australia Ltd	Buy	23,502	2,487	Ordinary
UBS Securities Australia Ltd Buy 1,465 155 UBS Securities Australia Ltd Buy 1,267 769 UBS Securities Australia Ltd Buy 1,267 769 UBS Securities Australia Ltd Buy 1,655 281 UBS Securities Australia Ltd Buy 1,655 281 UBS Securities Australia Ltd Buy 1,665 27 UBS Securities Australia Ltd Buy 1,666 42 UBS Securities Australia Ltd Buy 1,666 420 UBS Securities Australia Ltd Buy 1,064 1,065 UBS Securities Australia Ltd Buy 2,797 26 UBS Securities Australia Ltd Buy 1,797 26 UBS Securities Australia Ltd Buy 1,797 26 UBS Securities Australia Ltd Buy 1,797 26 UBS Securities Australia Ltd Buy 1,787 2,781 1,73 UBS Securities Australia Ltd Buy 1,787 2,781 1,73 UBS Securities Australia Ltd<		UBS Securities Australia Ltd	Buy	6,965	737	Ordinary
Buy 18,683 1,977 Buy 7,267 769 Buy 1,267 769 Buy 1,267 769 Buy 1,267 769 Buy 1,263 2,744 Buy 1,890 2,744 Buy 2,5,931 2,744 Buy 6,067 662 Buy 6,067 662 Buy 1,064 1,065 Buy 2,797 2,96 Buy 2,783 6,014 Buy 2,784 400 Buy 2,848 3,944 Buy 2,848 3,944 Buy 2,848 3,944 Buy 2,885 6,014 Buy 2,885 1,72 Buy 2,885 1,73 <td>11-Jul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>1.465</td> <td>155</td> <td>Ordinary</td>	11-Jul-14	UBS Securities Australia Ltd	Buy	1.465	155	Ordinary
UBS Securities Australia Ltd Buy 7,267 769 UBS Securities Australia Ltd Buy 7,267 769 UBS Securities Australia Ltd Buy 1,655 1,978 UBS Securities Australia Ltd Buy 1,655 1,978 UBS Securities Australia Ltd Buy 6,667 7,44 UBS Securities Australia Ltd Buy 6,667 6,07 UBS Securities Australia Ltd Buy 5,886 2,00 UBS Securities Australia Ltd Buy 5,886 2,00 UBS Securities Australia Ltd Buy 5,886 2,00 UBS Securities Australia Ltd Buy 5,886 3,00 UBS Securities Australia Ltd Buy 3,780 40 UBS Securities Australia Ltd Buy 3,176 1,23 UBS Securities Australia Ltd Buy 3,176 1,23 UBS Securities Australia Ltd Buy 3,176 1,23 UBS Securities Australia Ltd Buy 3,176 1,12 UBS Securities Australia Ltd Buy	l	UBS Securities Australia Ltd	Riiv	18 683	1 977	Crolings
UBS Securities Australia Ltd Buy 1.26 UBS Securities Australia Ltd Buy 2.655 281 UBS Securities Australia Ltd Buy 1.6890 1.978 UBS Securities Australia Ltd Buy 1.6890 1.978 UBS Securities Australia Ltd Buy 1.0664 1.978 UBS Securities Australia Ltd Buy 6.067 6.42 UBS Securities Australia Ltd Buy 6.087 6.071 UBS Securities Australia Ltd Buy 5.6832 6.014 UBS Securities Australia Ltd Buy 2.797 2.96 UBS Securities Australia Ltd Buy 2.797 2.96 UBS Securities Australia Ltd Buy 2.797 2.96 UBS Securities Australia Ltd Buy 2.835 3.04 UBS Securities Australia Ltd Buy 2.835 3.034 UBS Securities Australia Ltd Buy 2.835 6.21 UBS Securities Australia Ltd Buy 2.835 6.21 UBS Securities Australia Ltd Buy 2.835		188 Securities Australia i to	Busy	730.7	032	(Billian)
UBS Securities Australia Ltd Buy 2,655 281 UBS Securities Australia Ltd Buy 2,655 231 UBS Securities Australia Ltd Buy 2,697 1,578 UBS Securities Australia Ltd Buy 2,697 6,07 UBS Securities Australia Ltd Buy 6,087 6,07 UBS Securities Australia Ltd Buy 5,832 6,014 UBS Securities Australia Ltd Buy 5,882 6,014 UBS Securities Australia Ltd Buy 5,882 6,014 UBS Securities Australia Ltd Buy 5,887 5,806 UBS Securities Australia Ltd Buy 5,4867 5,806 UBS Securities Australia Ltd Buy 1,486 1,33 UBS Securities Australia Ltd Buy 1,247 1,25 UBS Securities Australia Ltd Buy 2,381 1,27 UBS Securities Australia Ltd Buy 2,381 1,39 UBS Securities Australia Ltd Buy 2,381 1,12 UBS Securities Australia Ltd Buy <td>L</td> <td>I ISO Cocurition Ametralia Lea</td> <td>500</td> <td>102,1</td> <td>60/</td> <td>Crainary</td>	L	I ISO Cocurition Ametralia Lea	500	102,1	60/	Crainary
UBS Securities Australia Ltd Buy 1,890 2081 UBS Securities Australia Ltd Buy 1,890 1,978 UBS Securities Australia Ltd Buy 6,067 6,42 UBS Securities Australia Ltd Buy 6,067 6,067 UBS Securities Australia Ltd Buy 6,067 6,067 UBS Securities Australia Ltd Buy 5,886 6,007 UBS Securities Australia Ltd Buy 2,797 2,96 UBS Securities Australia Ltd Buy 2,797 2,96 UBS Securities Australia Ltd Buy 1,547 2,06 UBS Securities Australia Ltd Buy 1,547 2,06 UBS Securities Australia Ltd Buy 1,537 3,034 UBS Securities Australia Ltd Buy 1,547 2,00 UBS Securities Australia Ltd Buy 1,784 1,53 UBS Securities Australia Ltd Buy 2,381 4,13 UBS Securities Australia Ltd Buy 2,381 4,13 UBS Securities Australia Ltd Buy <td></td> <td>Library Australia Ltd</td> <td>and a</td> <td>123</td> <td>5 5</td> <td>Ordinary</td>		Library Australia Ltd	and a	123	5 5	Ordinary
UBS Securities Australia Ltd Buy 1,8850 1,978 UBS Securities Australia Ltd Buy 1,8850 1,978 UBS Securities Australia Ltd Buy 6,67 647 UBS Securities Australia Ltd Buy 6,067 647 UBS Securities Australia Ltd Buy 6,067 640 UBS Securities Australia Ltd Buy 3,780 4,00 UBS Securities Australia Ltd Buy 3,780 4,00 UBS Securities Australia Ltd Buy 3,780 4,00 UBS Securities Australia Ltd Buy 3,780 2,00 UBS Securities Australia Ltd Buy 3,716 3,00 UBS Securities Australia Ltd Buy 3,716 3,00 UBS Securities Australia Ltd Buy 2,835 30 UBS Securities Australia Ltd Buy 2,835 1,75 UBS Securities Australia Ltd Buy 2,835 1,75 UBS Securities Australia Ltd Buy 2,835 1,736 UBS Securities Australia Ltd Buy	ı	COS Seculities Australia Liu	pay	2,655	187	Ordinary
UBS Securities Australia Ltd BMy 1,978 UBS Securities Australia Ltd Buy 5,631 7,44 UBS Securities Australia Ltd Buy 6,067 642 UBS Securities Australia Ltd Buy 6,087 6,067 642 UBS Securities Australia Ltd Buy 5,832 6,014 10,064 10,055 UBS Securities Australia Ltd Buy 2,798 6,014 10,055 UBS Securities Australia Ltd Buy 1,418 1,50 UBS Securities Australia Ltd Buy 1,424 1,53 UBS Securities Australia Ltd Buy 1,247 1,53 UBS Securities Australia Ltd Buy 2,381 3,934 UBS Securities Australia Ltd Buy 1,153 1,27 UBS Securities Australia Ltd Buy 2,381 1,395 UBS Securities Australia Ltd Buy 2,381 1,395 UBS Securities Australia Ltd Buy 2,381 1,395 UBS Securities Australia Ltd Buy 2,582 2,596	1	Ubs securities Australia Ltd	Buy	068'1	200	Ordinary
UBS Securities Australia Ltd Buy 5,331 2,744 UBS Securities Australia Ltd Buy 6,067 6,07 UBS Securities Australia Ltd Buy 1,066 400 UBS Securities Australia Ltd Buy 3,780 400 UBS Securities Australia Ltd Buy 3,780 400 UBS Securities Australia Ltd Buy 3,780 400 UBS Securities Australia Ltd Buy 1,847 206 UBS Securities Australia Ltd Buy 1,847 206 UBS Securities Australia Ltd Buy 1,625 172 UBS Securities Australia Ltd Buy 1,625 172 UBS Securities Australia Ltd Buy 1,746 153 UBS Securities Australia Ltd Buy 1,446 153 UBS Securities Australia Ltd Buy 2,811 2,521 UBS Securities Australia Ltd Buy 4,130 4,7 UBS Securities Australia Ltd Buy 4,130 4,7 UBS Securities Australia Ltd Buy 6	ł	UBS Securities Australia Ltd	Buy	18,692	1,978	Ordinary
UBS Securities Australia Ltd Buy 6,067 642 UBS Securities Australia Ltd Buy 6,067 642 UBS Securities Australia Ltd Buy 1,068 1,006 UBS Securities Australia Ltd Buy 2,797 2.96 UBS Securities Australia Ltd Buy 2,797 2.96 UBS Securities Australia Ltd Buy 2,797 2.06 UBS Securities Australia Ltd Buy 1,627 2.06 UBS Securities Australia Ltd Buy 3,7176 3,934 UBS Securities Australia Ltd Buy 1,625 172 UBS Securities Australia Ltd Buy 1,786 189 UBS Securities Australia Ltd Buy 2,881 620 UBS Securities Australia Ltd Buy 2,881 620 UBS Securities Australia Ltd Buy 4,130 1,153 1,00 UBS Securities Australia Ltd Buy 2,62,39 694 44 47 UBS Securities Australia Ltd Buy 2,62,39 694 1,00	1	UBS Securities Australia Ltd	Buy	25,931	2,744	Ordinary
UBS Securities Australia Ltd Bby 6.067 642 UBS Securities Australia Ltd Buy 10,064 1,065 UBS Securities Australia Ltd Buy 3,780 400 UBS Securities Australia Ltd Buy 2,797 2,96 UBS Securities Australia Ltd Buy 1,487 1,50 UBS Securities Australia Ltd Buy 1,487 1,50 UBS Securities Australia Ltd Buy 3,716 3,934 UBS Securities Australia Ltd Buy 3,716 3,934 UBS Securities Australia Ltd Buy 1,786 153 UBS Securities Australia Ltd Buy 1,786 153 UBS Securities Australia Ltd Buy 1,153 1,22 UBS Securities Australia Ltd Buy 2,381 4,7 UBS Securities Australia Ltd Buy 2,381 8,9 UBS Securities Australia Ltd Buy 4,130 4,7 UBS Securities Australia Ltd Buy 2,623 2,76 UBS Securities Australia Ltd Buy	١	UBS Securities Australia Ltd	Buy	99	7	Ordinary
UBS Securities Australia Ltd Buy 10,664 1,055 UBS Securities Australia Ltd Buy 56,832 6,014 UBS Securities Australia Ltd Buy 3,780 2,00 UBS Securities Australia Ltd Buy 2,797 2,66 UBS Securities Australia Ltd Buy 5,4687 5,806 UBS Securities Australia Ltd Buy 5,4867 5,806 UBS Securities Australia Ltd Buy 37,176 3,934 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 2,835 4,7 UBS Securities Australia Ltd Buy 4,130 4,7 UBS Securities Australia Ltd Buy 4,130 4,7 UBS Securities Australia Ltd Buy 8,48 8,44 UBS Securities Australia Ltd Buy	-14	UBS Securities Australia Ltd	Buy	6,067	642	Ordinary
UBS Securities Australia Ltd Buy 56,832 6,014 UBS Securities Australia Ltd Buy 2,790 400 UBS Securities Australia Ltd Buy 2,797 296 UBS Securities Australia Ltd Buy 1,418 1,50 UBS Securities Australia Ltd Buy 1,627 2,06 UBS Securities Australia Ltd Buy 3,716 3,934 UBS Securities Australia Ltd Buy 2,885 300 UBS Securities Australia Ltd Buy 2,881 252 UBS Securities Australia Ltd Buy 1,786 189 UBS Securities Australia Ltd Buy 2,381 252 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 2,6,233 2,776 UBS Securities Australia Ltd Buy 2,6,233 2,776 UBS Securities Australia Ltd Buy 2,6,233 2,776 UBS Securities Australia Ltd Buy	-14	UBS Securities Australia Ltd	Buy	10,064	1,065	Ordinary
UBS Securities Australia Ltd Buy 2,780 400 UBS Securities Australia Ltd Buy 2,797 256 UBS Securities Australia Ltd Buy 1,437 206 UBS Securities Australia Ltd Buy 1,547 206 UBS Securities Australia Ltd Buy 37,176 3,934 UBS Securities Australia Ltd Buy 37,176 3,934 UBS Securities Australia Ltd Buy 3,785 300 UBS Securities Australia Ltd Buy 1,783 2,281 UBS Securities Australia Ltd Buy 1,784 153 UBS Securities Australia Ltd Buy 1,784 153 UBS Securities Australia Ltd Buy 4,130 4,7 UBS Securities Australia Ltd Buy 4,130 4,7 UBS Securities Australia Ltd Buy 4,130 4,3 UBS Securities Australia Ltd Buy 8,48 8,9 UBS Securities Australia Ltd Buy 2,236 10 UBS Securities Australia Ltd Buy	-14	UBS Securities Australia Ltd	Buy	56,832	6,014	Ordinary
UBS Securities Australia Ltd Buy 2,797 2.96 UBS Securities Australia Ltd Buy 1,418 150 UBS Securities Australia Ltd Buy 5,467 5,066 UBS Securities Australia Ltd Buy 1,625 172 UBS Securities Australia Ltd Buy 2,831 300 UBS Securities Australia Ltd Buy 2,881 152 UBS Securities Australia Ltd Buy 2,881 122 UBS Securities Australia Ltd Buy 1,786 133 UBS Securities Australia Ltd Buy 1,153 122 UBS Securities Australia Ltd Buy 4,130 4,130 UBS Securities Australia Ltd Buy 4,130 4,130 UBS Securities Australia Ltd Buy 4,130 4,130 UBS Securities Australia Ltd Buy 26,233 2,766 UBS Securities Australia Ltd Buy 2,289 94 UBS Securities Australia Ltd Buy 2,289 94 UBS Securities Australia Ltd Buy	-14	UBS Securities Australia Ltd	Buv	3.780	400	Ordinary
UBS Securities Australia Ltd Buy 1,418 1,50 UBS Securities Australia Ltd Buy 5,4667 5,806 UBS Securities Australia Ltd Buy 1,627 1,58 UBS Securities Australia Ltd Buy 2,835 304 UBS Securities Australia Ltd Buy 2,835 304 UBS Securities Australia Ltd Buy 1,786 122 UBS Securities Australia Ltd Buy 1,146 123 UBS Securities Australia Ltd Buy 1,153 122 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 1,153 1,105 UBS Securities Australia Ltd Buy 2,6,233 2,706 UBS Securities Australia Ltd Buy 2,6,233 2,706 UBS Securities Australia Ltd Buy 2,6,233 2,706 UBS Securities Australia Ltd Buy 3,506 3,71 UBS Securities Australia Ltd Buy <td>ı</td> <td>UBS Securities Australia Ltd</td> <td>Biv</td> <td>2,707</td> <td>200</td> <td>Coding</td>	ı	UBS Securities Australia Ltd	Biv	2,707	200	Coding
UBS Securities Australia Ltd Buy 5,478 b UBS Securities Australia Ltd Buy 1,947 206 UBS Securities Australia Ltd Buy 1,623 206 UBS Securities Australia Ltd Buy 37,776 3,934 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 2,836 189 UBS Securities Australia Ltd Buy 2,836 153 UBS Securities Australia Ltd Buy 4,130 47 UBS Securities Australia Ltd Buy 8,48 894 UBS Securities Australia Ltd Buy 3,505 371 <td></td> <td>I IBS Securities Australia I to</td> <td>- A</td> <td>1 418</td> <td>257</td> <td>Cidinary</td>		I IBS Securities Australia I to	- A	1 418	257	Cidinary
UBS Securities Australia Ltd Buy 1,947 2,000 UBS Securities Australia Ltd Buy 1,625 172 UBS Securities Australia Ltd Buy 2,835 3034 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 2,881 252 UBS Securities Australia Ltd Buy 5,889 620 UBS Securities Australia Ltd Buy 4,130 4,130 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 26,239 94 UBS Securities Australia Ltd Buy 26,239 94 UBS Securities Australia Ltd Buy 26,239 94 UBS Securities Australia Ltd Buy 3,259 94 UBS Securities Australia Ltd Buy	Н	I IBC Constitute Australia I to	D	1,410	061	Orginary
UBS Securities Australia Ltd Buy 1,947 205 UBS Securities Australia Ltd Buy 37,176 3,934 UBS Securities Australia Ltd Buy 37,176 3,934 UBS Securities Australia Ltd Buy 1,885 300 UBS Securities Australia Ltd Buy 1,486 153 UBS Securities Australia Ltd Buy 1,153 122 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 1,53 1,76 UBS Securities Australia Ltd Buy 26,233 2,76 UBS Securities Australia Ltd Buy 26,233 2,76 UBS Securities Australia Ltd Buy 8,48 894 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy <	1	Upo Securities Australia Ltd	anà.	24,867	908,4	Ordinary
UBS Securities Australia Ltd Bby 1,625 172 UBS Securities Australia Ltd Bby 3,176 3,934 UBS Securities Australia Ltd Buy 2,885 300 UBS Securities Australia Ltd Buy 2,881 189 UBS Securities Australia Ltd Buy 2,889 620 UBS Securities Australia Ltd Buy 4,130 47 UBS Securities Australia Ltd Buy 6,20 95 UBS Securities Australia Ltd Buy 2,229 984 UBS Securities Australia Ltd Buy 2,229 984 UBS Securities Australia Ltd Buy 2,229 984 UBS Securities Australia Ltd Buy 3,206 371 UBS Securities Australia Ltd Buy 3,206	1	UBS Securities Australia Ltd	Buy	1,947	506	Ordinary
UBS Securities Australia Ltd Bby 3,934 UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 1,786 188 UBS Securities Australia Ltd Buy 1,746 153 UBS Securities Australia Ltd Buy 5,859 6,20 UBS Securities Australia Ltd Buy 4,150 1,75 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 1,54 1,0 UBS Securities Australia Ltd Buy 2,6,23 2,76 UBS Securities Australia Ltd Buy 2,6,23 2,76 UBS Securities Australia Ltd Buy 2,26,23 2,76 UBS Securities Australia Ltd Buy 2,29 964 UBS Securities Australia Ltd Buy 3,505 371 UBS Securities Australia Ltd Buy 3,505 371 UBS Securities Australia Ltd Buy 3,505 <	-	UBS Securities Australia Ltd	Buy	1,625	172	Ordinary
UBS Securities Australia Ltd Buy 2,835 300 UBS Securities Australia Ltd Buy 1,786 189 UBS Securities Australia Ltd Buy 2,811 252 UBS Securities Australia Ltd Buy 5,859 620 UBS Securities Australia Ltd Buy 4,130 477 UBS Securities Australia Ltd Buy 4,130 47 UBS Securities Australia Ltd Buy 4,130 47 UBS Securities Australia Ltd Buy 13,183 1,786 UBS Securities Australia Ltd Buy 8,48 894 UBS Securities Australia Ltd Buy 8,48 894 UBS Securities Australia Ltd Buy 8,289 994 UBS Securities Australia Ltd Buy 3,206 994 UBS Securities Australia Ltd Buy 3,506 974 UBS Securities Australia Ltd Buy 3,506 974 UBS Securities Australia Ltd Buy 3,502 974 UBS Securities Australia Ltd Buy 3,512 </td <td>- 1</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>37,176</td> <td>3,934</td> <td>Ordinary</td>	- 1	UBS Securities Australia Ltd	Buy	37,176	3,934	Ordinary
UBS Securities Australia Ltd Buy 1,786 198 UBS Securities Australia Ltd Buy 2,381 25.2 UBS Securities Australia Ltd Buy 1,446 620 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 47 UBS Securities Australia Ltd Buy 4,130 47 UBS Securities Australia Ltd Buy 4,130 95 UBS Securities Australia Ltd Buy 8,448 894 UBS Securities Australia Ltd Buy 2,229 984 UBS Securities Australia Ltd Buy 8,448 894 UBS Securities Australia Ltd Buy 2,239 984 UBS Securities Australia Ltd Buy 3,206 371 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,506 <td>- 1</td> <td>UBS Securities Australia Etd</td> <td>Buy</td> <td>2,835</td> <td>300</td> <td>Ordinary</td>	- 1	UBS Securities Australia Etd	Buy	2,835	300	Ordinary
UBS Securities Australia Ltd Bby 2,381 252 UBS Securities Australia Ltd Buy 1,446 153 UBS Securities Australia Ltd Buy 1,159 122 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 1,1,183 1,395 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 8,448 8,94 UBS Securities Australia Ltd Buy 2,529 964 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,505 414 UBS Securities Australia Ltd Buy		UBS Securities Australia Ltd	Buy	1,786	189	Ordinary
UBS Securities Australia Ltd Buy 1,446 153 UBS Securities Australia Ltd Buy 5,859 620 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 95 10 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 8,48 894 UBS Securities Australia Ltd Buy 6,299 964 UBS Securities Australia Ltd Buy 9,299 964 UBS Securities Australia Ltd Buy 3,506 31 UBS Securities Australia Ltd Buy 3,506 31 UBS Securities Australia Ltd Buy 3,502 414 UBS Securities Australia Ltd Buy 3,512 414 UBS Securities Australia Ltd Buy 3,512 414 UBS Securities Australia Ltd Buy 3,612 414 UBS Securities Australia Ltd Buy 3,612		UBS Securities Australia Ltd	Buy	2,381	252	Ordinary
UBS Securities Australia Ltd Buy 5,859 620 UBS Securities Australia Ltd Buy 1,153 122 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 444 47 UBS Securities Australia Ltd Buy 95 10 UBS Securities Australia Ltd Buy 26,233 2,766 UBS Securities Australia Ltd Buy 26,239 984 UBS Securities Australia Ltd Buy 2,299 984 UBS Securities Australia Ltd Buy 3,289 984 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,505 371 UBS Securities Australia Ltd Buy 3,505 371 UBS Securities Australia Ltd Buy 3,505 <td>-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>1.446</td> <td>153</td> <td>Ordinary</td>	-14	UBS Securities Australia Ltd	Buv	1.446	153	Ordinary
UBS Securities Australia Ltd Buy 1,153 122 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4430 47 UBS Securities Australia Ltd Buy 95 10 UBS Securities Australia Ltd Buy 26,233 2,766 UBS Securities Australia Ltd Buy 26,233 2,766 UBS Securities Australia Ltd Buy 8,488 894 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,512 414 UBS Securities Australia Ltd Buy 3,612 414 UBS Securities Australia Ltd Buy 3,612 414 UBS Securities Australia Ltd Buy 3,612 414 UBS Securities Australia Ltd Buy 3,612<	-14	UBS Securities Australia Ltd	Buv	5,859	620	Ordinary
UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 4,130 437 UBS Securities Australia Ltd Buy 56,233 1,00 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 2,239 894 UBS Securities Australia Ltd Buy 9,299 984 UBS Securities Australia Ltd Buy 1,229 130 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3591 414 UBS Securities Australia Ltd Buy 359 99	-14	UBS Securities Australia Ltd	Buv	1 153	122	Ordinary
USS Securities Australia Ltd Buy 444 47 UBS Securities Australia Ltd Buy 444 47 UBS Securities Australia Ltd Buy 26,233 1,76 UBS Securities Australia Ltd Buy 26,233 2,76 UBS Securities Australia Ltd Buy 8,448 894 UBS Securities Australia Ltd Buy 1,229 984 UBS Securities Australia Ltd Buy 1,229 130 UBS Securities Australia Ltd Buy 3,506 371	14	UBS Securities Australia I to	Buy	V 130	757	Ordinary Ordinary
USS Securities Australia Ltd Buy 13.82 10.95 UBS Securities Australia Ltd Buy 13.82 13.95 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 8,448 894 UBS Securities Australia Ltd Buy 3,290 994 UBS Securities Australia Ltd Buy 3,590 13.00 UBS Securities Australia Ltd Buy 3,590 13.00 UBS Securities Australia Ltd Buy 3,590 14.1 UBS Securities Australia Ltd Buy 3,912 414 UBS Securities Australia Ltd Buy 3,912 UBS	-	185 Societies Australia 14d	100	444	Ì	Cidinaly
UBS Securities Australia Ltd Buy 13,183 1,395 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 9,299 984 UBS Securities Australia Ltd Buy 1,229 130 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,506 414 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 359 99	1	I IBS Securities Australia (14)	in a	100	, t	Cidinary
UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 26,233 2,776 UBS Securities Australia Ltd Buy 9,289 984 UBS Securities Australia Ltd Buy 1,229 130 UBS Securities Australia Ltd Buy 3,506 3,71 UBS Securities Australia Ltd Buy 3,912 414 UBS Securities Australia Ltd Buy 3,912 UBS Securities Australia Ltd B		I IDC Committee Australia I tol	, and	50,50	2 5	Oldinary
USS Securities Australia Ltd Buy 8,488 894 UBS Securities Australia Ltd Buy 9,299 964 UBS Securities Australia Ltd Buy 1,229 964 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,912 414	14	IIRS Securities Australia 14d	Bus	567.30	377.0	Ordinary
UBS Securities Australia Ltd Buy 5,448 894 UBS Securities Australia Ltd Buy 9,299 94 UBS Securities Australia Ltd Buy 1,229 130 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,912 414 UBS Securities Australia Ltd Buy 339 98 UBS Securities Australia Ltd Buy 336 99	: :	1 DO Con without Australia Lea	, and	20,233	2,7,0	Orginary
UBS Securities Australia Ltd Buy 9,289 984 UBS Securities Australia Ltd Buy 1,229 130 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,912 414 UBS Securities Australia Ltd Buy 3,912 414 UBS Securities Australia Ltd Buy 3,912 414 UBS Securities Australia Ltd Buy 392 68	1	Ubs securities Australia Ltd	Buy	8,448	894	Ordinary
UBS Securities Australia Ltd Buy 1,229 130 UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,912 414 UBS Securities Australia Ltd Buy 936 99	4	UBS Securities Australia Ltd	Buy	9,299	984	Ordinary
UBS Securities Australia Ltd Buy 3,506 371 UBS Securities Australia Ltd Buy 3,512 414 UBS Securities Australia Ltd Buy 832 88 UBS Securities Australia Ltd Buy 936 99	14	UBS Securities Australia Ltd	Buy	1,229	130	Ordinary
UBS Securities Australia BW 3.912 414 UBS Securities Australia BW 832 88 UBS Securities Australia BW 935 99 UBS Securities Australia BW 936 99	14	UBS Securities Australia Ltd	Buy	3,506	371	Ordinary
UBS Securities Australia Ltd Buy 832 88 UBS Securities Australia Ltd Buy 936 99	14	UBS Securities Australia Ltd	Buy	3,912	414	Ordinary
UBS Securities Australia Ltd Buy 936 99	14	UBS Securities Australia Ltd	Buy	832	88	Ordinary
לסמי הביתו וונבי שתפו שוום דית		DE Counties Australia Ltd		9000	3 8	
	1	UBS Securities Australia Liu	Buy	9356	66	Ordinary

10-Jul-14 10-Jul-14 10-Jul-14	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	20,309	2,140	Ordinary
-Jul-14 -Jul-14	UBS Securities Australia Ltd	A de	7.2	4	Ordinary
-Jul-14		Loop L	,,	0	
	UBS Securities Australia Ltd	Buy	4,085	430	Ordinary
0-701-14	UBS Securities Australia Ltd	Buy	7,106	748	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buv	13.101	1,379	Ordinary
15.11.17	I IRS Securities Australia 14d	Britis	32 876	au/	Ording
	I DC Commission Australia 14-3	(no	ביסים	200	Ciciliary
11.00.0	U.D.C. Commission Australia Ltd	(ng	320,207	33,706	Ordinary
1000	Libe contines Australia Ltd	àna d	066,21	006,	Collidiy
-101-	Obs securities Australia Liu	pon	5,928	674	Ordinary
0-Jul-14	UBS Securities Australia Ltd	BUJ	5,18/	240	Ordinary
J-Jul-14	UBS Securities Australia Ltd	Bu <u>y</u>	3,078	324	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	5,643	594	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	2,849	300	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	38.684	4.072	Ordinary
0-tul-14	LIBS Securities Australia Ltd	Birv	38 503	4.036	Ordinary
0-hil-14	I IBS Securities Australia Ltd	Rus	5, 209	546	Ordinary
1	Libo Convision Australia Ltd		1050	:	diam'r.
10-201-14	The Commission Augustical and	ána c	600'1	- 000	Countries
- In-	ODS Securities Australia Ltd	ng	90,000	2,960	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	2,862	300	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	45,019	4,719	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	49,751	5,215	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	17,248	1,808	Ordinary
10-Jul-14	UBS Securities Australia Ltd	Buy	4,102	430	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	4,207	441	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	5,209	246	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	6)75	546	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	40,965	4,294	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	1.040	100	Ordinary
0.1.1.14	I IDC Con without Ametralia 14d	,,,,,	13 005	1 467	Ording
2 2 2	LIBS Committee Australia 144	500	25,21	1002.	Coloniary
-) Jac-	ODS Securities Australia Ltd	, and	16,210	3,	Cidinary
10-Jul-14	UBS Securities Australia Ltd	Buy	8,700	911	Ordinary
10-Jut-14	UBS Securities Australia Ltd	Buy	38	4	Ordinary
0-Jul-14	UBS Securities Australia Ltd	Buy	38	4	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	2,380	252	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buy	12,570	1,331	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buv	47.350	5.000	Ordinary
11-hil-14	I IBS Securities Australia I td	(Sell	106 514	(11 241)	Ordinary
1 1 1 1 1	1 10C Convision Aurenalia Indi		150,051	100.01	in in it
1-Jul-14	ODS Securities Australia Liu	ono.	109,901	500,0	Ordinary
-701-14	ODS Seculities Australia LIG	ána	40,577	4,333	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Sell	172,181	(18,225)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	216,185	(22,954)	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buy	415,787	44,027	Ordinary
1-Jul-14	UBS Wealth Management Australia Ltd	Sell	17.681	(1,869)	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buv	44,225	4.670	Ordinary
1-111-14	I IBS Securities Australia I to	- All	50 822	5317	Ordinan
	The state of the s	, and	22,042	100	Guinary
1-JUI-14	UBS Securities Australia Ltd	Buy	23,779	2,511	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	32,776	3,461	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	12,122	1,280	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	22,425	2,368	Ordinary
1-lui-14	UBS Securities Australia Ltd	Buy	11 998	1 267	Ordinary
1-1-4-14	I IBC Consulties Assertalia 14d	D. C.	20,521	2 167	Ordinan
1	COO OCCUMENT OUT IN THE	ĥ	120,02	4,107	o o
-Jui-14	OBS Securities Australia Ltd	Buy	,,033	one one	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	7,301	771	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	89,065	9,405	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buv	12,785	1,350	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buv	10.853	1.146	Ordinary
1-1-14	1 IRS Securities Australia 11d	Bris	737	32	Ordinary
1 1 1 1 1	COO Securities Australia Lid	600	107 c	2 2	Ordinary
-Jul-14	Obs securities Australia Ltd	had	5,703	50	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	19,186	2,026	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	49,954	5,275	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buv	388	41	Ordinary
-11-1-14	I IRS Securities Australia I to	218	909	64	Ordinary
1 24	Line Committee According and	San A	101000	12.000	Crainary Codings
)-Jul-14	UBS Securities Australia Ltd	Buy	121,869	12,869	Ordinary
1-Jul-14	LIBS Sequenties Australia 14	-	EE 441 1		Ordinani
		edy	144,00	200,0	Orginary
11-Jul-14	UBS Securities Australia Ltd	Buy	134,550	14,208	Ordinary

V	Obs Secondines Australia Liu	2	200	×	Crdinary
1-Jul-14	UBS Securities Australia Ltd	Buv	120	æ	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	11,191	1.183	Ordinary
Jul-14	UBS Securities Australia Ltd	Bliv	2 384	252	Ordinary
1-3 if-14	LIRS Securities Australia Ltd	B	200,2	767	Oldinary
1 1 1 1 7	I IBC Con without A section 144	500	074	t 6	Ordinary
	I IDC Commission A section of the	Sod.	740	60	Orginary
177	1 IBC Contribion Australia Ltd	èna o	858'I	502	Ordinary
	I DO COUNTIES AUSTRAIN LIN	ng	202	77	Ordinary
1-1-1-14	I IBC Committee Australia 144	śna	99/	S 8	Ordinary
	DDS Secontiles Australia cio	book	769'1	2002	Ordinary
41-JUI-14	Obs securities Australia Ltd	Buy	65,22/	6,895	Ordinary
- M- 14	USS Securities Australia Ltd	Buy	4,768	204	Ordinary
- JOI - 14	UBS Securities Australia Ltd	Buy	359	38	Ordinary
- Id	UBS Securities Australia Ltd	Buy	369	36	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	29,600	3,129	Ordinary
14-14	UBS Securities Australia Ltd	Buy	7,918	837	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	2,960	630	Ordinary
1-Jui-14	UBS Securities Australia Ltd	Buy	17,558	1,856	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	2,866	303	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	67,440	7,129	Ordinary
Jul-14	UBS Securities Australia Ltd	Buy	946	100	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	12,809	1,354	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	738	78	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	101,156	10,693	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	47	5	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buy	5,288	559	Ordinary
Jul-14	UBS Securities Australia Ltd	Buy	2,185	231	Ordinary
Jul-14	UBS Securities Australia Ltd	Buy	3,680	389	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buv	576	19	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	76	α	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buv	1.455	154	Ordinary
1-101-14	UBS Securities Australia I to	Bits	809 0	37.0	ordinary.
Jul-14	URS Securities Australia 11d	Bis	475	7/2	Ordinary
11-4-17	I IDC Con without Australia 144	600	624	Ç.	Ordinary
1 1 1 1 1	1 OC Counting Australia Ltd	Buy	4,328	458	Ordinary
1	COS Securities Australia Co	Buy	787	99	Ordinary
1 1	UBS Securities Australia Ltd	Buy	069	73	Ordinary
1-101-14	UBS Securities Australia Ltd	Buy	18,541	1,962	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	15,895	1,682	Ordinary
Jul-14	UBS Securities Australia Ltd	Buy	4,243	449	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	425	45	Ordinary
1-301-14	UBS Securities Australia Ltd	Buy	069	73	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	227	24	Ordinary
Jul-14	UBS Securities Australia Ltd	Buy	206	96	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buv	3.364	356	Ordinary
Jul-14	UBS Securities Australia Ltd	Birv	2,835		ording.
1-11-14	I JBS Securities Australia I td	Bir	2,032	315	Ordinary
1-hi-14	IRS Securities Australia Ital	(c) a	17.575	2000	Ordinary
1 1 1	TIPS Countries Australia Ltd	bus,	12,323	1,304	Orginary
1	I SO Constitute Australia 144	Sed.	4,004	430	Ordinary
1 1 1	li 100 Cocumition Acception to	bay b	09/15	400	Ordinary
1 1	Unit classical des Australia Lita	, and	500	8	Ordinary
1 - 17	ODS Securities Australia Ltd	Buy	614	65	Ordinary
1-101-14	Ubs securities Australia Ltd	Buy	6,832	723	Ordinary
1-JUI-14	UBS Securities Australia Ltd	Buy	1,956	202	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	5,670	900	Ordinary
Jul-14	UBS Securities Australia Ltd	Buy	2,608	276	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buy	340	36	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	40,427	4,278	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buy	3,780	400	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	9,450	1,000	Ordinary
lul-14	UBS Securities Australia Ltd	Buv	2.268	240	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	4.271	452	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buv	1,909	202	Ordinary
l	UBS Securities Australia Ltd	Buv	069	73	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buv	482	2 2	Ordinan,
ı					2000
	II IRC Contrition Attestial to	. ≥ 8	4 215	346	Ordinary

UBS Securities Australia Ltd Buy UBS Securities Australia Ltd Buy	1-1-14	Obs securities Australia LIQ				Ordinary
Miles Securities Australia Ltd Buy 1,782 1,783 4,93 Miles Securities Australia Ltd Buy 1,782 1,783 4,93 Miles Securities Australia Ltd Buy 1,782 1,783 1,783 Miles Securities Australia Ltd Buy 1,782 1,783 1,783 Miles Securities Australia Ltd Buy 1,782 1,783 1,783 Miles Securities Australia Ltd Buy 1,783 1,783 1,783 Miles Securities Australia Ltd Buy 1,783 1,783 1,783 1,783 Miles Securities Australia Ltd Buy 1,783 1,783 1,783 1,783 Miles Securities Australia Ltd Buy 1,783	1.1.14	I IBS Sacurities Australia 14d	ouy Our	304	1	
(8) Securities Australia Ltd 60 y 4,723 450 (8) Securities Australia Ltd 60 y 4,723 450 (8) Securities Australia Ltd 60 y 5,727 770 (8) Securities Australia Ltd 60 y 7,227 770 (8) Securities Australia Ltd 60 y 7,227 770 (8) Securities Australia Ltd 60 y 7,227 720 (8) Securities Australia Ltd 60 y 7,227 720 (8) Securities Australia Ltd 60 y 7,274 720 (8) Securities Australia Ltd 60 y 7,274 720 (8) Securities Australia Ltd 60 y 7,275 720 (8) Securities Australia Ltd 60 y 7,274 720 (8) Securities Australia Ltd 60 y 7,275 720 (8) Securities Australia Ltd 60 y 7,275 720 (8) Securities Australia Ltd 60 y 7,725 720 (8) Securities Australia Ltd 60 y 7,725 720 (8) Securities Australia Ltd 60 y	1	Upo Counties Australia tid	Sing.	7,144	95/	Ordinary
68 Securities Australia Ltd Buy 5,737 770 68 Securities Australia Ltd Buy 5,782 770 68 Securities Australia Ltd Buy 5,982 2,603 68 Securities Australia Ltd Buy 5,982 2,003 68 Securities Australia Ltd Buy 7,405 7,70 68 Securities Australia Ltd Buy 6,993 7,70 10 Securities Australia Ltd Buy 6,903 7,70 10 Securities Australia Ltd Buy 6,503 2,503 10 Securities Australia Ltd Buy 6,503 2,503 10 Securities Australia Ltd Buy 6,503 3,503 10 Securities Australia Ltd Buy 7,523 7,504 10 Securities Australia Ltd Buy 7,523 7,504 10 Securities Australia Ltd Buy 7,	1	Table Committee Acetalia Ltd	euy Pres	3,780	400	Orginary
(BS Securities Australia Ltd BM 7,277 770 548 (BS Securities Australia Ltd BM 5,892 653 650 (BS Securities Australia Ltd BM 5,894 650 650 (BS Securities Australia Ltd BM 5,894 650 650 (BS Securities Australia Ltd BM 5,894 7,271 700 (BS Securities Australia Ltd BM 7,894 7,80 771 700 (BS Securities Australia Ltd BM 1,153 7,73 700 700 (BS Securities Australia Ltd BM 6,00 7,73 700 700 (BS Securities Australia Ltd BM 6,00 7,73 700 700 (BS Securities Australia Ltd BM 6,00 7,73 700 700 (BS Securities Australia Ltd BM 6,00 7,73 700 700 (BS Securities Australia Ltd BM 7,73 7,73 7,73 7,74 (BS Securities Australia Ltd BM 7,73	1 1 1	Obs seculines Australia Liu	Buy	4,253	450	Ordinary
UBS Securities Australia Ltd BM 5,527 770 UBS Securities Australia Ltd BM 5,984 6,59 UBS Securities Australia Ltd BM 5,984 6,50 UBS Securities Australia Ltd BM 7,40 7,40 UBS Securities Australia Ltd BM 7,40 7,20 UBS Securities Australia Ltd BM 7,40 7,20 UBS Securities Australia Ltd BM 7,70 7,70 UBS Securities Australia Ltd BM 7,70 <t< td=""><td>10-17</td><td>Obs securities Australia Ltd</td><td>buy</td><td>9,1,9</td><td>548</td><td>Ordinary</td></t<>	10-17	Obs securities Australia Ltd	buy	9,1,9	548	Ordinary
UBS Securities Autorials Ltd Buy 5,982 630 UBS Securities Autorials Ltd Buy 5,982 630 UBS Securities Autorials Ltd Buy 5,982 630 UBS Securities Autorials Ltd Buy 5,982 724 UBS Securities Autorials Ltd Buy 5,983 727 UBS Securities Autorials Ltd Buy 6,933 727 UBS Securities Autorials Ltd Buy 6,933 728 UBS Securities Autorials Ltd Buy 6,733 651 UBS Securities Autorials Ltd Buy 7,732 7,83 UBS Securities Autorials Ltd Buy 7,732 7,93 UBS Securities Autorials Ltd Buy 7,732 7,73 UBS Securities Autorials Ltd Buy 7,25 7,23 UBS Securities Autorials Ltd Buy	4-14	UBS Securities Australia Ltd	Buy	7,277	077	Ordinary
UBS Securities Australia Ltd Buy 5,954 2.00 UBS Securities Australia Ltd Buy 18,000 2.00 UBS Securities Australia Ltd Buy 2,000 3,445 3,655 UBS Securities Australia Ltd Buy 2,000 3,655 3,700 UBS Securities Australia Ltd Buy 2,700 3,700 3,700 UBS Securities Australia Ltd Buy 3,700 3,700 3,700 UBS Securities Australia Ltd Buy 3,700 3,700 3,700 UBS Securities Australia Ltd Buy 3,700 3,700 3,700 3,700 UBS Securities Australia Ltd Buy 5,700 3,70	JUI-14	UBS Securities Australia Ltd	Buy	5,982	633	Ordinary
UBS Securities Australia Ltd Buy 18,358 3,645 UBS Securities Australia Ltd Buy 3,445 3,645 UBS Securities Australia Ltd Buy 6,533 1,22 LBS Securities Australia Ltd Buy 6,533 1,22 LBS Securities Australia Ltd Buy 4,561 2,55 LBS Securities Australia Ltd Buy 4,567 3,48 LBS Securities Australia Ltd Buy 5,50 3,59 LBS Securities Australia Ltd Buy 5,50 3,50 LBS Securities Australia Ltd Buy 5,50 3,50 LBS Securities Australia Ltd Buy 5,50 3,50 LBS Securities Australia Ltd Buy 6,50 7,51 LBS Securities Australia Ltd Buy 6,50 7,52 LBS Securities Australia Ltd Buy 8,50 7,52 LBS Securities Australia Ltd Buy 8,50 7,52 LBS Securities Australia Ltd Buy 2,50 7,52 LBS Securities Australia Ltd Buy <	Jul-14	UBS Securities Australia Ltd	Buy	5,954	630	Ordinary
UBS Securities Australia Ltd Buy 34,445 3,645 UBS Securities Australia Ltd Buy 2,72 10.2 UBS Securities Australia Ltd Buy 2,73 17.0 UBS Securities Australia Ltd Buy 4,561 17.2 UBS Securities Australia Ltd Buy 4,561 37.7 UBS Securities Australia Ltd Buy 4,573 3,48 UBS Securities Australia Ltd Buy 4,73 3,78 UBS Securities Australia Ltd Buy 4,73 79 UBS Securities Australia Ltd Buy 4,73 79 UBS Securities Australia Ltd Buy 5,73 79 UBS Securities Australia Ltd Buy 5,74 7,73 UBS Securities Australia Ltd Buy 5,74 7,73 UBS Securities Australia Ltd Buy 7,27	Jul-14	UBS Securities Australia Ltd	Buy	18,928	2,003	Ordinary
UBS Securities Australia Ltd Buy 274 28 UBS Securities Australia Ltd Buy 6 591 740 UBS Securities Australia Ltd Buy 4,561 2.55 UBS Securities Australia Ltd Buy 4,563 3.73 UBS Securities Australia Ltd Buy 3,673 3.49 UBS Securities Australia Ltd Buy 3,736 3.249 UBS Securities Australia Ltd Buy 3,736 3.50 UBS Securities Australia Ltd Buy 3,736 3.61 UBS Securities Australia Ltd Buy 3,737 3.62 UBS Securities Australia Ltd Buy 3,737 3.62 UBS Securities Australia Ltd Buy 3,737 3.63 UBS Securities Australia Ltd Buy 3,738 3.73 UBS Securities Australia Ltd Buy 2,838 3.73 UBS Securities Australia Ltd Buy 2,838 3.73 UBS Securities Australia Ltd Buy 2,838 3.73 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	34,445	3,645	Ordinary
UBS Securities Australia Ltd Bay 6.993 740 UBS Securities Australia Ltd Buy 4.153 7.12 UBS Securities Australia Ltd Buy 4.563 3.74 UBS Securities Australia Ltd Buy 4.75 3.28 UBS Securities Australia Ltd Buy 4.70 3.73 UBS Securities Australia Ltd Buy 4.70 3.74 UBS Securities Australia Ltd Buy 4.70 3.78 UBS Securities Australia Ltd Buy 4.70 3.78 UBS Securities Australia Ltd Buy 4.75 4.70 UBS Securities Australia Ltd Buy 4.75 4.70 UBS Securities Australia Ltd Buy 4.75 4.70 UBS Securities Australia Ltd Buy 2.73 4.71 UBS Securities Australia Ltd Buy 2.72	Jul-14	UBS Securities Australia Ltd	Buy	274	20	Ordinary
UBS Securities Australia Ltd Buy 1,155 1,25 UBS Securities Australia Ltd Buy 3,568 3,491 UBS Securities Australia Ltd Buy 3,578 3,497 UBS Securities Australia Ltd Buy 30,702 3,590 UBS Securities Australia Ltd Buy 4,702 4,500 UBS Securities Australia Ltd Buy 4,207 1,500 UBS Securities Australia Ltd Buy 5,204 1,500 UBS Securities Australia Ltd Buy 2,504 1,500 UBS Securities Australia Ltd Buy 2,504 2,504 UBS Securities Australia Ltd	Jul-14	UBS Securities Australia Ltd	Biv	5003	OPZ	Ordinary
UBS Securities Australia Ltd Buy 4,551 377 UBS Securities Australia Ltd Buy 4,552 377 UBS Securities Australia Ltd Buy 4,752 3,543 UBS Securities Australia Ltd Buy 5,73 6,53 6,51 UBS Securities Australia Ltd Buy 7,73 7,73 1,90 UBS Securities Australia Ltd Buy 7,73 1,90 1,90 UBS Securities Australia Ltd Buy 7,73 1,90 1,90 UBS Securities Australia Ltd Buy 7,73 1,90 1,90 UBS Securities Australia Ltd Buy 6,669 7,03 1,90 UBS Securities Australia Ltd Buy 6,669 7,03 1,90 UBS Securities Australia Ltd Buy 6,669 7,03 1,19 UBS Securities Australia Ltd Buy 6,669 7,03 1,19 UBS Securities Australia Ltd Buy 7,50 2,53 6,669 1,19 UBS Securities Australia Ltd Buy 7,669	11-14	I IBS Securities Australia I td	Bux	1159	56.1	Ordinal y
USS Securities Australia Ltd Buy 3,581 37.7 USS Securities Australia Ltd Buy 3,782 34.9 USS Securities Australia Ltd Buy 30,702 34.9 USS Securities Australia Ltd Buy 30,702 34.9 USS Securities Australia Ltd Buy 30,702 3.24 USS Securities Australia Ltd Buy 30,702 3.24 USS Securities Australia Ltd Buy 4,705 3.78 USS Securities Australia Ltd Buy 4,705 4.70 USS Securities Australia Ltd Buy 4,705 1,70 USS Securities Australia Ltd Buy 4,705 1,70 USS Securities Australia Ltd Buy 4,705 1,70 USS Securities Australia Ltd Buy 6,600 1,70 USS Securities Australia Ltd Buy 6,600 1,73 USS Securities Australia Ltd Buy 1,78 1,73 USS Securities Australia Ltd Buy 1,78 1,73 USS Securities Australia Ltd Buy	1	I IDC Committee American Inch	6.6	501,1	77.	o della
UBS Securities Australia Ltd Buy 3,593 497 UBS Securities Australia Ltd Buy 4,703 324 UBS Securities Australia Ltd Buy 5,253 651 UBS Securities Australia Ltd Buy 5,253 651 UBS Securities Australia Ltd Buy 3,493 790 UBS Securities Australia Ltd Buy 4,26 1,285 UBS Securities Australia Ltd Buy 3,487 1,146 UBS Securities Australia Ltd Buy 3,487 1,158 UBS Securities Australia Ltd Buy 3,684 1,158 UBS Securities Australia Ltd Buy 3,784 1,158 UBS Securities Australia Ltd Buy 3,784 1,158 UBS Securities Australia Ltd Buy 2,520 1,158 UBS Securities Australia Ltd Buy 2,520 1,158 UBS Securities Australia Ltd Buy 2,322 1,242 UBS Securities Australia Ltd Buy 2,322 2,243 UBS Securities Australia Ltd Buy	100	Opposed the Australia and	, and	4,30	575	Ordinary
UBS Securities Australia Ltd Bluy 30,702 348 UBS Securities Australia Ltd Bluy 30,702 348 UBS Securities Australia Ltd Bluy 30,702 348 UBS Securities Australia Ltd Bluy 4,702 348 UBS Securities Australia Ltd Bluy 4,737 790 UBS Securities Australia Ltd Bluy 4,257 1,504 UBS Securities Australia Ltd Bluy 4,257 1,504 UBS Securities Australia Ltd Bluy 4,257 4,504 UBS Securities Australia Ltd Bluy 6,669 7,05 UBS Securities Australia Ltd Bluy 6,669 7,05 UBS Securities Australia Ltd Bluy 7,527 4,237 1,00 UBS Securities Australia Ltd Bluy 7,527 4,237 1,00 UBS Securities Australia Ltd Bluy 7,527 4,237 1,00 UBS Securities Australia Ltd Bluy 7,527 1,00 1,00 UBS Securities Australia Ltd Bluy 7,527	100	Coo seculities Australia Liu	euy	3,563	3//	Ordinary
UBS Securities Australia Ltd Bluy 90,736 3,249 UBS Securities Australia Ltd Bluy 573 661 UBS Securities Australia Ltd Bluy 6253 661 UBS Securities Australia Ltd Bluy 7,473 790 UBS Securities Australia Ltd Bluy 9,466 1,000 UBS Securities Australia Ltd Bluy 3,784 4,000 UBS Securities Australia Ltd Bluy 2,393 1,154 UBS Securities Australia Ltd Bluy 2,343 3,00 UBS Securities Australia Ltd B	# - I	UBS Securities Australia Ltd	Buy	4,702	497	Ordinary
UBS Securities Australia Ltd Bluy 520 651 UBS Securities Australia Ltd Bluy 6.235 651 UBS Securities Australia Ltd Bluy 4.25 7.473 7.90 UBS Securities Australia Ltd Bluy 4.25 1.000 1.000 UBS Securities Australia Ltd Bluy 4.257 1.89 1.1,256 1.100 UBS Securities Australia Ltd Bluy 4.257 4.00 1.00 UBS Securities Australia Ltd Bluy 4.257 4.00 UBS Securities Australia Ltd Bluy 5.89 5.784 4.00 UBS Securities Australia Ltd Bluy 5.89 5.73 4.00 UBS Securities Australia Ltd Bluy 5.89 5.73 4.00 UBS Securities Australia Ltd Bluy 5.89 5.73 4.23 1.05 UBS Securities Australia Ltd Bluy 5.80 5.73 4.23 1.05 UBS Securities Australia Ltd Bluy 5.78 7.43 1.05 UBS Securities Australia	Jul-14	UBS Securities Australia Ltd	Buy	30,736	3,249	Ordinary
UBS Securities Australia Ltd Bluy 6.553 661 UBS Securities Australia Ltd Bluy 369 39 UBS Securities Australia Ltd Bluy 369 425 UBS Securities Australia Ltd Bluy 426 1,000 UBS Securities Australia Ltd Bluy 9,466 1,000 UBS Securities Australia Ltd Bluy 9,466 1,000 UBS Securities Australia Ltd Bluy 3,784 4,00 UBS Securities Australia Ltd Bluy 3,784 4,00 UBS Securities Australia Ltd Bluy 5,20 7,05 UBS Securities Australia Ltd Bluy 2,30 3,0 UBS Securities Australia Ltd Bluy 2,30 3,0 UBS Securities Australia Ltd Bluy 2,30 6,6 UBS Securities Australia Ltd Bluy 3,12 3,0 UBS Securities Australia Ltd Bluy 3,12 3,0 UBS Securities Australia Ltd Bluy 3,12 3,0 UBS Securities Australia Ltd Bluy	3UI-14	UBS Securities Australia Ltd	Buy	520	55	Ordinary
UBS Securities Australia Ltd Buy 366 39 UBS Securities Australia Ltd Buy 7,473 793 UBS Securities Australia Ltd Buy 7,473 793 UBS Securities Australia Ltd Buy 4,260 1,000 UBS Securities Australia Ltd Buy 8,425 1,136 UBS Securities Australia Ltd Buy 8,425 1,136 UBS Securities Australia Ltd Buy 1,284 400 UBS Securities Australia Ltd Buy 1,584 1,160 UBS Securities Australia Ltd Buy 2,565 705 UBS Securities Australia Ltd Buy 2,303 1,734 UBS Securities Australia Ltd Buy 2,303 1,734 1,50 UBS Securities Australia Ltd Buy 2,304 1,72 2,33 UBS Securities Australia Ltd Buy 2,344 3,53 UBS Securities Australia Ltd Buy 2,344 3,53 UBS Securities Australia Ltd Buy 2,34 3,74 UBS Securities Au	Jul-14	UBS Securities Australia Ltd	Buy	6,253	199	Ordinary
UBS Securities Australia Ltd Buy 478 789 UBS Securities Australia Ltd Buy 426 4.00 UBS Securities Australia Ltd Buy 9,460 1,000 UBS Securities Australia Ltd Buy 1,396 1,134 UBS Securities Australia Ltd Buy 1,396 1,134 UBS Securities Australia Ltd Buy 1,262 1,000 UBS Securities Australia Ltd Buy 5,669 705 UBS Securities Australia Ltd Buy 5,669 705 UBS Securities Australia Ltd Buy 5,869 705 UBS Securities Australia Ltd Buy 5,869 1,869 UBS Securities Australia Ltd Buy 2,302 2,244 UBS Securities Australia Ltd Buy 2,302 2,244 UBS Securities Australia Ltd Buy 7,542 1,887 UBS Securities Australia Ltd Buy 7,542 1,887 UBS Securities Australia Ltd Buy 7,542 1,887 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	369	39	Ordinary
UBS Securities Australia Ltd Buy 426 1,00 UBS Securities Australia Ltd Buy 9,460 1,00 UBS Securities Australia Ltd Buy 8,457 1,89 UBS Securities Australia Ltd Buy 1,381 1,46 UBS Securities Australia Ltd Buy 1,296 7,11 UBS Securities Australia Ltd Buy 4,257 450 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 2,338 300 UBS Securities Australia Ltd Buy 2,344 103 UBS Securities Australia Ltd Buy 2,344 165 UBS Securities Australia Ltd Buy 1,344 1,324 168 UBS Securities Australia Ltd Buy 1,344 1,324 1,444 UBS Securities Austral	Jul-14	UBS Securities Australia Ltd	Buy	7,473	790	Ordinary
UBS Securities Australia Ltd Buy 9,4460 1,000 UBS Securities Australia Ltd Buy 8,457 8,94 UBS Securities Australia Ltd Buy 1,396 1,194 UBS Securities Australia Ltd Buy 1,396 7,05 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 2,830 6,67 73 UBS Securities Australia Ltd Buy 2,830 6,67 73 UBS Securities Australia Ltd Buy 2,830 6,67 73 UBS Securities Australia Ltd Buy 7,782 1,846 1,92 UBS Securities Australia Ltd Buy 7,782 1,846 1,92 UBS Securities Australia Ltd Buy 7,782	Jul-14	UBS Securities Australia Ltd	Buy	426	45	Ordinary
UBS Securities Australia Ltd Buy 8,457 894 UBS Securities Australia Ltd Buy 1,381 1,46 UBS Securities Australia Ltd Buy 1,381 1,143 UBS Securities Australia Ltd Buy 1,596 2,11 UBS Securities Australia Ltd Buy 6,689 705 UBS Securities Australia Ltd Buy 6,699 705 UBS Securities Australia Ltd Buy 2,320 6,434 UBS Securities Australia Ltd Buy 7,744 1,69 UBS Securities Australia Ltd Buy 7,744 1,60 UBS Securities Australia Ltd Buy 7,744 1,64 UBS Securities Australia Ltd Buy 7,744 1,64 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	9,460	1,000	Ordinary
UBS Securities Australia Ltd Buy 1,181 1,184 UBS Securities Australia Ltd Buy 1,186 1,187 UBS Securities Australia Ltd Buy 2,784 400 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 4,257 450 UBS Securities Australia Ltd Buy 4,257 450 UBS Securities Australia Ltd Buy 2,3026 2,434 UBS Securities Australia Ltd Buy 1,842 1,866 UBS Securities Australia Ltd Buy 2,74 2,93 UBS Securities Australia Ltd Buy 2,74 2,93 UBS Securities Australia Ltd Buy 2,54 2,50 UBS Securities Australia Ltd Buy 2,54 2,50 UBS Securities Australia Ltd Buy	Jul-14	UBS Securities Australia Ltd	Buy	8 457	708	Ordinary
UBS Securities Australia Ltd Buy 11,295 1,194 UBS Securities Australia Ltd Buy 1,696 211 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 6,838 300 UBS Securities Australia Ltd Buy 6,838 300 UBS Securities Australia Ltd Buy 6,338 300 UBS Securities Australia Ltd Buy 2,3026 2,434 UBS Securities Australia Ltd Buy 2,3026 2,434 UBS Securities Australia Ltd Buy 2,3026 2,434 UBS Securities Australia Ltd Buy 2,34 3,15 UBS Securities Australia Ltd Buy 3,14 3,29 UBS Securities Australia Ltd Buy 3,14 3,29 UBS Securities Australia Ltd Buy 3,14 1,84 UBS Securities Australia Ltd Buy 3,14 3,29 UBS Securities Australia Ltd Buy	11-14	UBS Securities Australia I to	Bix	1381	146	Ordinan
UBS Securities Australia Ltd Buy 1,926 1,175 UBS Securities Australia Ltd Buy 1,926 211 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 4,257 430 UBS Securities Australia Ltd Buy 2,838 300 UBS Securities Australia Ltd Buy 2,838 300 UBS Securities Australia Ltd Buy 2,838 300 UBS Securities Australia Ltd Buy 2,330 131 UBS Securities Australia Ltd Buy 2,330 136 UBS Securities Australia Ltd Buy 1,7,842 1,886 UBS Securities Australia Ltd Buy 1,7,842 1,886 UBS Securities Australia Ltd Buy 1,7,842 1,844 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 3,144 1,844 UBS Securities Australia Ltd Buy 2,583 2,09 UBS Securities Australia Ltd Buy	11-14	I IBS Securities Australia I td	Buy	11 305	,	Ordings
USE Securities Australia Ltd Buy 6.659 400 USE Securities Australia Ltd Buy 6.669 705 USE Securities Australia Ltd Buy 6.669 705 USE Securities Australia Ltd Buy 6.669 705 USE Securities Australia Ltd Buy 2.338 3.00 USE Securities Australia Ltd Buy 2.938 3.00 USE Securities Australia Ltd Buy 2.933 1.160 USE Securities Australia Ltd Buy 2.93 1.18 USE Securities Australia Ltd Buy 2.93 1.18 USE Securities Australia Ltd Buy 2.94 1.92 USE Securities Australia Ltd Buy 2.74 2.93 USE Securities Australia Ltd Buy 1.7444 1.844 USE Securities Australia Ltd Buy 3.124 3.29 USE Securities Australia Ltd Buy 3.124 3.29 USE Securities Australia Ltd Buy 2.583 2.73 USE Securities Australia Ltd Buy	11-14	I IBS Securities Australia I to	7.0	9001	1	Carling C
UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 2,838 300 UBS Securities Australia Ltd Buy 2,934 1,160 UBS Securities Australia Ltd Buy 2,934 1,160 UBS Securities Australia Ltd Buy 2,934 1,160 UBS Securities Australia Ltd Buy 2,942 1,86 UBS Securities Australia Ltd Buy 2,74 1,92 UBS Securities Australia Ltd Buy 3,744 1,84 UBS Securities Australia Ltd Buy 2,58 2,66 UBS Securities Australia Ltd Buy	17.17	Labor Constitution Assertation Lead		V66,1	1 000	Oldinary
UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 4,257 450 UBS Securities Australia Ltd Buy 6,669 705 UBS Securities Australia Ltd Buy 2,838 300 UBS Securities Australia Ltd Buy 2,3026 2,434 UBS Securities Australia Ltd Buy 1,7842 1,886 UBS Securities Australia Ltd Buy 1,7842 1,886 UBS Securities Australia Ltd Buy 1,744 1,844 UBS Securities Australia Ltd Buy 1,744 1,844 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,583 2,78 UBS Securities Australia Ltd Bu		COO Securities Australia Ltd	ed.	5,704	400	Ordinary
UBS Securities Australia Ltd Buy 6.69 705 UBS Securities Australia Ltd Buy 6.69 705 UBS Securities Australia Ltd Buy 6.69 705 UBS Securities Australia Ltd Buy 23,026 2,434 UBS Securities Australia Ltd Buy 23,026 2,434 UBS Securities Australia Ltd Buy 29,03 31 UBS Securities Australia Ltd Buy 29,3 31 UBS Securities Australia Ltd Buy 274 29 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,13 329 UBS Securities Australia Ltd Buy 3,13 329 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 2,583 2,66 UBS Securities Australia Ltd Buy 2,563	1	Obs securities Australia Ltd	gnà	699'9	,05	Ordinary
UBS Securities Australia Ltd Bluy 6,669 705 UBS Securities Australia Ltd Buy 2,838 300 UBS Securities Australia Ltd Buy 2,932 131 UBS Securities Australia Ltd Buy 2,934 1,160 UBS Securities Australia Ltd Buy 2,93 1,31 UBS Securities Australia Ltd Buy 609 606 64 UBS Securities Australia Ltd Buy 1,816 1,82 UBS Securities Australia Ltd Buy 7,93 43 UBS Securities Australia Ltd Buy 7,93 43 UBS Securities Australia Ltd Buy 7,93 84 UBS Securities Australia Ltd Buy 7,93 84 UBS Securities Australia Ltd Buy 7,53 182 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 2,543 1,724 1,844 UBS Securities Australia Ltd Buy 2,583 2,66 UBS Securities Australia Ltd	Jul-14	UBS Securities Australia Ltd	Buy	4,257	450	Ordinary
UBS Securities Australia Ltd Buy 1,542 163 UBS Securities Australia Ltd Buy 2,038 300 UBS Securities Australia Ltd Buy 29,03 1,160 UBS Securities Australia Ltd Buy 29,73 1,160 UBS Securities Australia Ltd Buy 29,3 1,160 UBS Securities Australia Ltd Buy 29,3 1,18 UBS Securities Australia Ltd Buy 27,4 29 UBS Securities Australia Ltd Buy 407 23 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 3,124 1,644 UBS Securities Australia Ltd Buy 1,7444 1,644 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,5	IUI-14	UBS Securities Australia Ltd	Buy	699'9	202	Ordinary
UBS Securities Australia Ltd Buy 2,838 300 UBS Securities Australia Ltd Buy 23,026 2,434 UBS Securities Australia Ltd Buy 10,974 1,160 UBS Securities Australia Ltd Buy 974 105 UBS Securities Australia Ltd Buy 974 105 UBS Securities Australia Ltd Buy 974 108 UBS Securities Australia Ltd Buy 274 29 UBS Securities Australia Ltd Buy 274 29 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,132 329 UBS Securities Australia Ltd Buy 1,724 184 UBS Securities Australia Ltd Buy 1,724 184 UBS Securities Australia Ltd Buy 1,744 1,84 UBS Securities Australia Ltd Buy 2,516 206 UBS Securities Australia Ltd Buy 2,516 206 UBS Securities Australia Ltd Buy 2,516	Iul-14	UBS Securities Australia Ltd	Buy	1,542	163	Ordinary
UBS Securities Australia Ltd Buy 691 73 UBS Securities Australia Ltd Buy 23,026 2,434 UBS Securities Australia Ltd Buy 10,974 1,105 UBS Securities Australia Ltd Buy 605 64 UBS Securities Australia Ltd Buy 1,7842 1,886 UBS Securities Australia Ltd Buy 1,7842 1,886 UBS Securities Australia Ltd Buy 492 52 UBS Securities Australia Ltd Buy 492 52 UBS Securities Australia Ltd Buy 7,73 182 UBS Securities Australia Ltd Buy 3,112 3.29 UBS Securities Australia Ltd Buy 3,124 3.29 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,583 2,78 UBS Securities Australia Ltd Buy	Iul-14	UBS Securities Australia Ltd	Buy	2,838	300	Ordinary
UBS Securities Australia Ltd Buy 23,026 2,434 UBS Securities Australia Ltd Buy 1,160 1,160 UBS Securities Australia Ltd Buy 574 1,160 UBS Securities Australia Ltd Buy 574 1,03 UBS Securities Australia Ltd Buy 574 1,886 UBS Securities Australia Ltd Buy 274 29 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,134 329 UBS Securities Australia Ltd Buy 3,144 1,844 UBS Securities Australia Ltd Buy 1,7444 1,844 UBS Securities Australia Ltd Buy 2,583 69 UBS Securities Australia Ltd Buy 2,583 69 UBS Securities Australia Ltd Buy 2,583 69 UBS Securities Australia Ltd Buy 2,583	11-14	UBS Securities Australia Ltd	Buy	109	27	Ordinan
UBS Securities Australia Ltd Buy 10,572 1,760 UBS Securities Australia Ltd Buy 2,721 1,160 UBS Securities Australia Ltd Buy 605 64 UBS Securities Australia Ltd Buy 1,784 1,886 UBS Securities Australia Ltd Buy 492 5,2 UBS Securities Australia Ltd Buy 492 5,2 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,434 363 UBS Securities Australia Ltd Buy 3,434 363 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,722 184 UBS Securities Australia Ltd Buy 2,583 256 UBS Securities Australia Ltd Buy 2,583 276 UBS Securities Australia Ltd Buy 2,583 276 UBS Securities Australia Ltd Buy 2,327 <td>Iul-14</td> <td>UBS Securities Australia Ltd</td> <td>Blav</td> <td>33 038</td> <td>NEN C</td> <td>Cacingo</td>	Iul-14	UBS Securities Australia Ltd	Blav	33 038	NEN C	Cacingo
UBS Securities Australia Ltd Buy 293 31 UBS Securities Australia Ltd Buy 293 31 UBS Securities Australia Ltd Buy 17,842 1.86 UBS Securities Australia Ltd Buy 1,816 192 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 2,548 1,04 UBS Securities Australia Ltd Buy 2,533 2,56 UBS Securities Australia Ltd Buy 2,327 2,46 UBS Securities Australia Ltd Buy 2,327 2,46 UBS Securities Australia Ltd Buy 2,327<	11-14	I IBS Securities Australia I td	Big	10.074	1160	Croins
UBS Securities Australia Ltd Buy 974 105 UBS Securities Australia Ltd Buy 974 105 UBS Securities Australia Ltd Buy 17,842 1,886 UBS Securities Australia Ltd Buy 274 29 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,132 329 UBS Securities Australia Ltd Buy 3,144 1,844 UBS Securities Australia Ltd Buy 1,724 1,84 UBS Securities Australia Ltd Buy 1,724 1,84 UBS Securities Australia Ltd Buy 1,724 1,84 UBS Securities Australia Ltd Buy 2,516 2,66 UBS Securities Australia Ltd Buy 2,516 2,66 UBS Securities Australia Ltd Buy 2,538 2,73 UBS Securities Australia Ltd Buy 2,538 2,66 UBS Securities Australia Ltd Buy 2,538 </td <td></td> <td>Library American by Library</td> <td>(p)</td> <td>1600</td> <td>20,10</td> <td>Countries</td>		Library American by Library	(p)	1600	20,10	Countries
UBS Securities Australia Ltd Buy 605 64 UBS Securities Australia Ltd Buy 17,842 1,886 UBS Securities Australia Ltd Buy 1,816 192 UBS Securities Australia Ltd Buy 492 52 UBS Securities Australia Ltd Buy 492 52 UBS Securities Australia Ltd Buy 795 84 UBS Securities Australia Ltd Buy 3,434 363 UBS Securities Australia Ltd Buy 3,434 363 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 2,516 65 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,237 2,66 UBS Securities Australia Ltd Buy 2,237 2,66 UBS Securities Australia Ltd Buy 2,237 <td>1 1</td> <td>Opp Securities Australia Liu</td> <td>euy</td> <td>567</td> <td>5</td> <td>Ordinary</td>	1 1	Opp Securities Australia Liu	euy	567	5	Ordinary
UBS Securities Australia Ltd Buy 17,842 1,86 UBS Securities Australia Ltd Buy 1,7842 1,86 UBS Securities Australia Ltd Buy 274 29 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,433 363 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 1,724 1,66 UBS Securities Australia Ltd Buy 2,518 273 UBS Securities Australia Ltd Buy 2,583 266 UBS Securities Australia Ltd Buy 2,583 266 UBS Securities Australia Ltd Buy 2,323 246 UBS Securities Australia Ltd Buy 2,323 246 UBS Securities Australia Ltd Buy 2,323 <td>10-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>9/4</td> <td>103</td> <td>Ordinary</td>	10-14	UBS Securities Australia Ltd	Buy	9/4	103	Ordinary
UBS Securities Australia Ltd Bby 17,842 1,886 UBS Securities Australia Ltd Buy 1,816 192 UBS Securities Australia Ltd Buy 274 29 UBS Securities Australia Ltd Buy 492 52 UBS Securities Australia Ltd Buy 795 84 UBS Securities Australia Ltd Buy 3,434 353 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 1,744 1,844 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,233 236 UBS Securities Australia Ltd Buy 2,237 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 <td>UF-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>909</td> <td>64</td> <td>Ordinary</td>	UF-14	UBS Securities Australia Ltd	Buy	909	64	Ordinary
UBS Securities Australia Ltd Buy 1,816 192 UBS Securities Australia Ltd Buy 497 52 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 2,516 269 UBS Securities Australia Ltd Buy 2,518 273 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,327 2,46 UBS Securities Australia Ltd Buy 2,327 2,46 UBS Securities Australia Ltd Buy 2,327 2,46 UBS Securities Australia Ltd Buy 2,3	U!-14	UBS Securities Australia Ltd	Buy	17,842	1,886	Ordinary
UBS Securities Australia Ltd Buy 274 29 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 795 84 UBS Securities Australia Ltd Buy 732 132 UBS Securities Australia Ltd Buy 8,637 913 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,237 246 UBS Securities Australia Ltd Buy 2,237 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 <	ui-14	UBS Securities Australia Ltd	Buy	1,816	192	Ordinary
UBS Securities Australia Ltd Bby 492 52 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 759 84 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,518 273 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,237 2,46 UBS Securities Australia Ltd Buy 2,327 2,46 UBS Securities Australia Ltd Buy 4,64 </td <td>Ul-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>774</td> <td>20</td> <td>Ordinary</td>	Ul-14	UBS Securities Australia Ltd	Buv	774	20	Ordinary
USS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 407 43 UBS Securities Australia Ltd Buy 3,434 363 UBS Securities Australia Ltd Buy 3,172 329 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,724 1,844 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,518 273 UBS Securities Australia Ltd Buy 2,583 69 UBS Securities Australia Ltd Buy 2,583 69 UBS Securities Australia Ltd Buy 2,583 236 UBS Securities Australia Ltd Buy 2,583 246 UBS Securities Australia Ltd Buy 2,323 246 UBS Securities Australia Ltd Buy 2,323 246 UBS Securities Australia Ltd Buy 2,323 246 UBS Securities Australia Ltd Buy 4,64	11-14	I IRS Securities Australia I to	Birk	2007	2 5	Ordinary
UBS Securities Australia Ltd Buy 795 84 UBS Securities Australia Ltd Buy 795 84 UBS Securities Australia Ltd Buy 3,132 359 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,722 184 UBS Securities Australia Ltd Buy 1,744 1,844 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,583 273 UBS Securities Australia Ltd Buy 2,583 276 UBS Securities Australia Ltd Buy 2,283 256 UBS Securities Australia Ltd Buy 2,237 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 464	1 1 1 1	Cop Securities Abstralla Lid	Buy	264	25	Ordinary
UBS Securities Australia Ltd Bluy 343 383 UBS Securities Australia Ltd Buy 3,413 329 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,742 1,644 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,518 273 UBS Securities Australia Ltd Buy 2,583 273 UBS Securities Australia Ltd Buy 2,583 274 UBS Securities Australia Ltd Buy 2,583 275 UBS Securities Australia Ltd Buy 2,323 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 4,64<	UI-14	UBS Securities Australia Ltd	Buy	407	43	Ordinary
UBS Securities Australia Ltd Buy 3,434 365 UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,744 1,844 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,516 104 UBS Securities Australia Ltd Buy 2,583 236 UBS Securities Australia Ltd Buy 2,333 236 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 464	ul-14	UBS Securities Australia Ltd	Buy	795	84	Ordinary
UBS Securities Australia Ltd Buy 3,112 329 UBS Securities Australia Ltd Buy 1,722 183 UBS Securities Australia Ltd Buy 1,744 1,844 UBS Securities Australia Ltd Buy 1,744 1,644 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,233 2,26 UBS Securities Australia Ltd Buy 2,327 2,46 UBS Securities Australia Ltd Buy 2,323 2,26 UBS Securities Australia Ltd Buy 2,323 2,46 UBS Securities Australia Ltd Buy 2,323 2,46 UBS Securities Australia Ltd Buy 2,327 2,46 UBS Securities Australia Ltd Buy 4,64 4,9 UBS Securities Australia Ltd Buy 7,152 7,56 UBS Securities Australia Ltd Buy	U-14	UBS Securities Australia Ltd	Buy	3.434	363	Ordinary
UBS Securities Australia Ltd Buy 8.637 913 UBS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,744 1,644 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,84 1,03 UBS Securities Australia Ltd Buy 2,83 273 UBS Securities Australia Ltd Buy 2,233 246 UBS Securities Australia Ltd Buy 2,233 246 UBS Securities Australia Ltd Buy 2,337 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7152 756 UBS Securities Australia Ltd Buy 464	U-14	UBS Securities Australia Ltd	Buy	3112	320	Ordinary
USS Securities Australia Ltd Buy 1,722 182 UBS Securities Australia Ltd Buy 1,722 184 UBS Securities Australia Ltd Buy 1,603 1,665 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,583 2,73 UBS Securities Australia Ltd Buy 2,283 77 UBS Securities Australia Ltd Buy 2,233 256 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 4,64 49 UBS Securities Australia Ltd Buy 4,64 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 7,152 <td>11.14</td> <td>I IBS Sociation Australia 14d</td> <td>2 6</td> <td>0 637</td> <td>200</td> <td>o di di</td>	11.14	I IBS Sociation Australia 14d	2 6	0 637	200	o di di
UBS Securities Australia Ltd Buy 17,424 182 UBS Securities Australia Ltd Buy 1,724 184 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 653 267 UBS Securities Australia Ltd Buy 2,583 273 UBS Securities Australia Ltd Buy 2,583 77 UBS Securities Australia Ltd Buy 2,533 265 UBS Securities Australia Ltd Buy 2,533 266 UBS Securities Australia Ltd Buy 2,533 246 UBS Securities Australia Ltd Buy 2,537 246 UBS Securities Australia Ltd Buy 2,5600 3,129 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,192 126 UBS Securities Australia Ltd Buy 7,152 126 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152		Liber Countries Assessed and	ĥ	/50'0	CIA	Cranary
UBS Securities Australia Ltd Buy 1,7444 1,844 UBS Securities Australia Ltd Buy 1,033 106 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 2,583 273 UBS Securities Australia Ltd Buy 984 104 UBS Securities Australia Ltd Buy 2,233 236 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152	1	Obs securities Australia Ltd	euy	77/1	781	Ordinary
UBS Securities Australia Ltd Buy 1.003 1.06 UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 653 69 UBS Securities Australia Ltd Buy 2,583 273 UBS Securities Australia Ltd Buy 728 77 UBS Securities Australia Ltd Buy 723 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 7,152	IUI-14	UBS Securities Australia Ltd	Buy	17,444	1,844	Ordinary
UBS Securities Australia Ltd Buy 2,516 266 UBS Securities Australia Ltd Buy 653 69 UBS Securities Australia Ltd Buy 2,683 27 UBS Securities Australia Ltd Buy 2,683 273 UBS Securities Australia Ltd Buy 2,233 273 UBS Securities Australia Ltd Buy 2,233 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 4,64 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 7,152	Iul-14	UBS Securities Australia Ltd	Buy	1,003	106	Ordinary
UBS Securities Australia Ltd Buy 653 69 UBS Securities Australia Ltd Buy 2,583 273 UBS Securities Australia Ltd Buy 984 104 UBS Securities Australia Ltd Buy 2,233 226 UBS Securities Australia Ltd Buy 2,233 226 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 4,64 49 UBS Securities Australia Ltd Buy 7,152 126 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 7,152	UI-14	UBS Securities Australia Ltd	Buy	2,516	266	Ordinary
UBS Securities Australia Ltd Buy 2,583 273 UBS Securities Australia Ltd Buy 984 104 UBS Securities Australia Ltd Buy 728 77 UBS Securities Australia Ltd Buy 2,233 276 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,9600 3,129 UBS Securities Australia Ltd Buy 1,192 126 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 9,152 756 UBS Securities Australia Ltd Buy 9,152 756 UBS Securities Australia Ltd Buy 132 14	Ul-14	UBS Securities Australia Ltd	Buv	653	69	Ordinary
UBS Securities Australia Ltd Buy 984 104 USS Securities Australia Ltd Buy 984 104 UBS Securities Australia Ltd Buy 2,233 236 UBS Securities Australia Ltd Buy 8,561 905 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 2,327 126 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 9,375 991 UBS Securities Australia Ltd Buy 3,152 991 UBS Securities Australia Ltd Buy 3,752 991 UBS Securities Australia Ltd Buy 3,752 991	U-14	UBS Securities Australia 11d	Bliv	2 583	273	Ordinan
USS Securities Australia Ltd Buy 728 104 UBS Securities Australia Ltd Buy 728 77 UBS Securities Australia Ltd Buy 2,233 2.56 UBS Securities Australia Ltd Buy 8,561 905 UBS Securities Australia Ltd Buy 2,327 2.46 UBS Securities Australia Ltd Buy 1,192 1.75 UBS Securities Australia Ltd Buy 4,64 49 UBS Securities Australia Ltd Buy 7,152 756		1 IRC Contrition Australia 14d	Birth	2,000	6,7	Cidinaly
UBS Securities Australia Ltd Buy 7.28 77 UBS Securities Australia Ltd Buy 2.233 2.56 UBS Securities Australia Ltd Buy 8.561 905 UBS Securities Australia Ltd Buy 2.327 246 UBS Securities Australia Ltd Buy 7.99 1.79 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7.152 756 UBS Securities Australia Ltd Buy 7.152 756 UBS Securities Australia Ltd Buy 9.375 991 UBS Securities Australia Ltd Buy 132 14	:	COO Securities Australia Liu	(ng	ton.	101	Ordinary
UBS Securities Australia Ltd Buy 2.233 236 UBS Securities Australia Ltd Buy 8,561 905 UBS Securities Australia Ltd Buy 2,327 2.46 UBS Securities Australia Ltd Buy 7,192 1.26 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 76	UI-14	UBS Securities Australia Ltd	Buy	728	77	Ordinary
UBS Securities Australia Ltd Buy 8,561 905 UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 29,600 3,129 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 137 14	ul-14	UBS Securities Australia Ltd	Buy	2,233	236	Ordinary
UBS Securities Australia Ltd Buy 2,327 246 UBS Securities Australia Ltd Buy 29,600 3,129 UBS Securities Australia Ltd Buy 1,192 1,78 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 9,152 756 UBS Securities Australia Ltd Buy 9,375 991 UBS Securities Australia Ltd Buy 132 14	ul-14	UBS Securities Australia Ltd	Buy	8,561	905	Ordinary
UBS Securities Australia Ltd Buy 29,600 3,129 UBS Securities Australia Ltd Buy 1,192 126 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 9,375 991 UBS Securities Australia Ltd Buy 132 14	Ul-14	UBS Securities Australia Ltd	Buv	2,327	246	Ordinary
USS Securities Australia Ltd Buy 5,750 1,75 UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 9,375 991 UBS Securities Australia Ltd Buy 132 14	11-14	ILIBS Securities Australia I to	Biv	29 FM	3 120	Ordinan
UBS Securities Australia Ltd Buy 1,192 126 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 9,375 991 UBS Securities Australia Ltd Buy 9,375 991 UBS Securities Australia Ltd Buy 132 14		CO COUNTY OF THE PARTY OF THE P	500	000,62	5,1,2	Gullary
UBS Securities Australia Ltd Buy 464 49 UBS Securities Australia Ltd Buy 7,152 756 UBS Securities Australia Ltd Buy 9,375 91 UBS Securities Australia Ltd Buy 132 14	N-14	UBS Securities Australia Ltd	Buy	1,192	126	Ordinary
UBS Securities Australia Ltd Buy 7,152 756 LUBS Securities Australia Ltd Buy 9,375 991 LUBS Securities Australia Ltd Buy 132 14	ul-14	UBS Securities Australia Ltd	Buy	464	49	Ordinary
UBS Securities Australia Ltd Buy 9,375 991 UBS Securities Australia Ltd Buy 132 14	ul-14	UBS Securities Australia Ltd	Buy	7,152	756	Ordinary
UBS Securities Australia Ltd Buy 132 14	ul-14	UBS Securities Australia Ltd	Buy	9,375	991	Ordinary
Upp Securities Australia Ltd Duy		I IDC Committee Assessed and	, ,			(200)
	1-14	UBS Securities Australia Ltd	200	((,	,	- 0

-101-14	I IDC Committee Acceptant				
1-1-1	UBS Securities Australia Ltd	Buy	416	444	Ordinary
1-10-14	UBS Securities Australia Itd	Buy	30,051	3,180	Ordinary
1-4-14	1 BS Societies Australia 14d	Buny	2,525	797	Orginary
1	ODS SECULIES AUSTRAIN LIG	pny	2,438	728	Ordinary
4 - 101 - 14	Obs securities Australia Ltd	Buy	616,	838	Ordinary
1 1	Opp Securities Australia Lid	pan)	OED, I	109	Ordinary
+ 1-101-	Coo Securities Australia Ltd	Buy	2,835	300	Ordinary
1 1 1 1	Obs Securities Australia Ltd	BUY	3,345	354	Ordinary
1 1 1 1 1	Libe Committee Australia Ltd	Buy	089,11	1,236	Ordinary
1 1 1 1	Upo Securities Australia Ltd	ána c	42,383	4,485	Ordinary
11-11-11	Libe cognition Australia Ltd	ing.	930	66	Ordinary
	Specialities Australia Liu	anà.	405	102	Ordinary
10.1	Obs Securities Australia Ltd	gnà	33,//4	3,5/4	Ordinary
+ In-In-	UBS Securities Australia Ltd	Buy	4,460	472	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	5,405	572	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	31,960	3,382	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	4,867	515	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	444	47	Ordinary
-Jui-14	UBS Securities Australia Ltd	Buy	1,285	136	Ordinary
IDI14	UBS Securities Australia Ltd	Buy	3,336	353	Ordinary
1-JUI-14	UBS Securities Australia Ltd	Buy	7,919	838	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	208	22	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	13,107	1,387	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buy	1,096	116	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	3,780	400	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	14,099	1,492	Ordinary
11-301-14	UBS Securities Australia Ltd	Buy	7,248	797	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	4,064	430	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	1,285	136	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	3,780	400	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buv	22.860	2.419	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buv	5 179	548	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Buv	2.268	240	Ordinary
11-Jul-14	UBS Securities Australia I td	BIN	28.350	0000	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	378	2,000	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Bity	6379	575	Ordinal y
-101-14	1 BS Securities Australia 14d	Bray	676	0,0	Ordinary
11 11 14	DDD Securities Australia Ltd	pon)	979	8	Ordinary
1 2	The Committee Australia Ltd	, and	125,654	13,220	Ordinary
	Upo Securities Australia Ltd	and	3,046	383	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	589,812	61,955	Ordinary
-1nr-1	UBS Securities Australia Ltd	Buy	11,900	1,250	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	29	3	Ordinary
11-Jui-14	UBS Securities Australia Ltd	Buy	29,921	3,143	Ordinary
-Jul-14	UBS Securities Australia Ltd	Buy	34,738	3,649	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Sell	25	(9)	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Sell	105	(11)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	660'9	(638)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	38	(4)	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Sell	621	(65)	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Seil	182	(61)	Ordinary
11-Jul-14	UBS Securities Australia Ltd	Sell	124	(13)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	64.616	(6.759)	Ordinary
-10-14	UBS Securities Australia I to	روا	27,370	(65,70)	Ordinary
-111-14	I IRS Securities Australia I to	Coll	070,12	(2,003)	Oldinary
11-10-17	LIDS Countries Australia Ltd	Jei	0/8	(16)	Orginary
1-1-1-17	LIBS Societifies Australia Ltd	lac lac	617,61	(255)	Ordinary
1-1-1-1-1	LIDS Securities Australia Ltd	llac 3	3,014	(3/8)	Ordinary
101.14	1 DC Committee Australia Ltd	Sell	/99′/1	(1,848)	Ordinary
1 1 1 1	Obs Securities Australia Ltd	Jes S	0/8	(16)	Ordinary
41-JUI-14	UBS Securities Australia Ltd	Seil	440	(46)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Set	33,651	(3,520)	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Sell	4,780	(200)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	10,373	(1,085)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	105	(11)	Ordinary
-Jul-14	UBS Securities Australia Ltd	Sell	53	(3)	Ordinary
11-jul-14	UBS Securities Australia Ltd	CAIL	30		
		-	5	5	Ordinary

	1,122 1,389 1,389 1,389 1,365 1,365 3,241 4,77 4,77 4,77 4,77 4,77 2,334 2,334 2,334 2,334 2,334 2,334 2,334 2,334 2,334 2,334 2,334 2,334 2,334 2,334 2,340 1,564	1,706 1,446 1,446 1,446 3,43 3,43 3,02 3,02 3,02 3,02 1,109 1	Ordinary Ordinary
	1,389 1,389 2,2,104 1,341 4,177 4,177 4,177 4,186 1,0,80 1,0,	2,339 1,446 1,446 1,446 1,446 1,446 1,446 1,446 1,446 1,446 1,446 1,146 1,	Ordinary Ordinary
	2,1,389 2,1,344 3,241 4,77 4,77 4,77 4,77 4,77 4,186 5,39 10,480 1,841 1,1,89 1,1,89 1,687 1,687 1,687 1,687 1,5819	2 339 1 446 1 1446 1 1446 1 1446 2 34 3 4 34 3 4 34 3 4 4 4 7 1 109 1	Ordinary Ordinary
	13,656 13,616 13,616 13,616 10,480 10,480 10,480 10,480 11,189 4,687 11,189 1,589 1,589 1,580 1,280	1,446 1,446 1,446 1,446 1,109 1,	Ordinary Ordinary
	3,202 3,202 4,177 4,177 2,884 2,334 2,334 2,334 2,334 2,334 2,339 3,24 1,189 4,86 6,870 6,870 6,870 1,189 1,189 4,86 1,189 1,189 1,189 1,189 1,280 2,362	1,440 3,430 1,507 1,007 1,00	Ordinary Ordinary
	4,77 4,77 4,77 2,884 2,884 2,884 2,884 4,186 6,870 10,480 18,418 11,189 1,687 1,687 1,519 1,	442 442 247 247 443 443 444 417 419 419 419 419 419 419 419 419	Ordinary Ordinary
	4,177 4,177 2,834 2,334 2,334 2,334 4,186 3,10480 10,480 10,480 11,189 4,687 11,189 4,687 1,580 1,580 1,580 1,580 1,295 1,580 1,295 1,2	422 442 547 777 777 777 777 777 767 767 7	Ordinary Ordinary
	2,884 2,334 2,634 2,334 2,634 2,639 3,175 2,640 11,189 11,	247 247 77 77 77 77 71,109 1,1	Ordinary Ordinary
	2,384 2,284 2,284 4,186 4,186 4,186 1,480 1,480 1,480 1,1,189 1,654 1,654 1,654 1,564 1,564 1,564 1,564 1,564 1,564 1,564 1,564 1,564 1,564 1,564 1,564 1,251 1,25	247 443 443 443 443 446 446 446 446	Ordinary Ordinary
	2,334 4,186 539 10,480 10,480 18,418 1,641 1,641 1,687 1,687 1,687 1,687 1,687 1,687 1,687 1,687 1,687 1,687 1,280 1	247 777 1,1097 1,1097 1,1097 1,1097 1,507 1,	Ordinary Ordinary
	728 4,186 1,186 1,186 837 6,870 1,1,189 1,1,189 1,564 1,564 1,564 1,654 1,2351 1,2351 1,2360 1,2,380 1,2,380 1,2,380 1,2,380 1,2,381 1,427 1,431 1,88	77 77 77 77 79 89 87 98 98 98 98 98 98 98 98 98 98 98 98 98	Ordinary Ordinary
	4,186 539 10,480 387 6,870 11,189 4,687 11,189 1,554 1,554 1,554 1,554 1,554 1,554 1,554 1,535 1,235 1	4477 1,109 1,109 1,184 1	Ordinary Ordinary
	539 539 10,480 87 6370 11,189 4,687 11,189 4,687 11,189 1,654	1,109 1,109 1,109 1,109 1,109 1,109 1,074 1,674	Ordinary Ordinary
	10,480 10,480 387 6,870 11,189 11,189 14,87 15,619 11,654 11,654 11,654 11,654 11,654 11,654 11,654 11,654 11,654 12,350 12,	1,109 1,109 1,249 1,184	Ordinary Ordinary
	10,480 887 6,870 18,418 5,037 11,189 4,687 1,684 1,684 1,684 1,684 1,280 12,361 13,81 14,18	1,109 1,294 1,294 1,844 1,184 1,184 1,674 1,674 1,674 1,674 1,874	Ordinary Ordinary
	387 6,870 18,418 11,189 4,687 1,5819 1,5819 1,5819 1,554 23,625 3,175 23,625 3,175 5,29 2,280 2,280 2,280 2,280 2,381 1,427 6,24 7,494 2,303 3,21 1,417 1,427 1,427 1,427 1,427 1,427 1,427 1,437 1,431 1,418	1,949 1,949 1,184 4,96 1,674 1	Ordinary Ordinary
	6,870 18,418 5,037 11,189 4,687 1,564 1,564 1,564 1,564 1,57 2,29 2,29 3,21 1,251 1,	1,249 1,349 1,184 1,184 1,187 1,187 1,187 1,27 1,307 1,307 1,307 1,307 1,511 1	Ordinary Ordinary
	18,478 1,189 4,687 1,5819 1,5819 1,581 2,362 3,07 1,235 1,235 1,235 1,235 1,235 2,39 2,39 2,39 2,39 2,39 2,39 1,235	1,949 1,184 1,1674 1,674	Ordinary Ordinary
	5,047 o 1,1,189 1,1,189 1,654 1,654 1,654 1,654 1,235 3,175 3,175 3,175 3,175 3,175 3,175 1,236 1,237 6,24 7,494 2,2,80 2,2,80 1,2,31 1,427 1,427 1,418 1,418	1,184 1,184	Ordinary Ordinary
	1,189 4,687 4,687 1,5819 1,664 1,664 23,625 3,175 3,175 3,175 3,175 3,175 3,175 3,175 3,175 3,175 3,175 3,175 1,280 1,280 2,380 2,380 2,380 2,380 2,380 2,380 2,48	1,184 496 1,674 1,674 1,877 2,500 2,500 1,307 1,307 1,307 1,307 1,307 1,307 1,307 1,507 1,	Ordinary Ordinary
	4,687 4,687 1,5819 1,654 23,625 3,175 3,175 5,88 2,288	1,844 1,674 1,674 1,500 3,500 3,500 1,307	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	4,687 1,5819 1,654 1,654 23,625 3,175 3,175 5,280 1,2,580 1,2,580 2,2,800 1,2,51 1,2,51 1,2,51 1,2,51 1,2,51 1,2,51 1,2,60 1,2,6	496 1,574 1,774 1,78 2,500 336 56 56 57 1,307 2,73 2,73 2,73 2,73 2,73 2,73 2,73 2,7	Ordinary Ordinary
	15,819 1,654 7,164 7,164 3,175 3,175 3,20 12,350 1,427 6,24 7,494 2,3,039 2,3,039 2,188 1,418	1,674 175 176 1,500 336 336 56 57 1,307 1,307 2,73 2,73 2,73 2,438 2,438 2,438	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	1,654 718 23,625 3,175 3,175 350 12,350 1,237 1,427 1,427 1,427 1,439 2,2309 3,21 2,309 3,21 1,418	76 76 76 76 76 336 336 56 56 73 73 73 73 73 73 73 73 73 73 73 73 73	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	23,625 3,175 3,175 3,175 3,50 12,350 1,2351 1,235 1,240 1,241 1,241 23,039 23,039 23,11 1,418 1,418	2,500 336 336 336 1,307 1,307 2,50 5,50 2,50 1,307 1,51 1,51 1,51 1,51 1,51 1,51 1,51 2,43 3,43 3,44 3,44 3,44 3,44 3,44 3,44	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	23,625 3,175 229 229 358 2,580 1,2,351 1,427 624 7,494 7,494 23,039 23,039 23,138 1,418	2,500 336 56 56 57 73 273 151 151 66 66 66 793 2,738 2	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	3,102 3,102 529 350 12,351 1,251 1,250 2,260	2,336 56 56 77 1,307 151 151 66 793 2,438 34	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	175 529 350 12,351 2,580 1,427 624 2,303 2,1 2,303 2,1 5,188 1,418	256 37 1,307 1,307 2,73 2,73 2,438 34 549	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	359 350 12,351 558 2,580 2,280 624 7,494 2,309 321 5,188 1,848 1,843	56 1,307 2,73 2,73 66 66 793 2,438 34 34	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	12,351 12,351 558 2,580 1,427 1,427 23,039 23,13 1,418 1,418	37 1,307 59 273 151 66 793 2,438 34	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	12,351 558 2,580 1,427 624 7,494 23,039 23,039 21,188 1,418	1,307 59 273 151 66 66 793 2,438 34	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	558 2,580 1,427 1,438 2,2309 23,11 5,188 1,418	273 273 151 66 793 2,438 34 549	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	2,580 1,427 624 23,039 21 2,188 1,8,18 1,418	273 151 66 793 2,438 34 549	Ordinary Ordinary Ordinary Ordinary
	1,427 624 7,494 23,039 321 5,188 1,884 1,418	151 66 793 2,438 34 549	Ordinary Ordinary Ordinary
	624 7,494 23,039 321 5,188 18,843 1,418	2,438 34 549	Ordinary Ordinary Ordinary
	7,494 23,039 321 5,188 18,843 14,18	793 2,438 34 549	Ordinary
	23,039 321 5,188 18,843 1,418	2,438	Ordinary
	5,188 18,843 1,418	34	Grannary
	5,188 18,843 1,418	549	
	5,188 18,843 1,418	243	Oldinary
	18,843		Ordinary
	1,418	1,994	Ordinary
		150	Ordinary
	23,219	2,457	Ordinary
	208	22	Ordinary
	350	ac	Creipary
	600	00	Cloriary
	41,807	4,474	Ordinary
	18,711	1,980	Ordinary
	501	23	Ordinary
	3,005	318	Ordinary
	24 485	2.591	Ordinary
	1 295	354	Ordingo
	002.6	200	Ordinary
	207,0	3	Clumary
	38/	41	Ordinary
	246	56	Ordinary
UBS Securities Australia Ltd	5,358	292	Ordinary
	10,017	1,060	Ordinary
	3,912	414	Ordinary
curities Australia Ltd	9.856	1 043	Ordinary
LIBS Securities Australia Ltd	25,241	2,671	Ordina
	010	100	in in in
	509.9	8 8	Ordinaly
	0,000	550	o di iai y
cutifies Australia Liu	4,462	45	Ordinary
	20,129	2,130	Ordinary
UBS Securities Australia Ltd	198	21	Ordinary
	482	51	Ordinary
UBS Securities Australia Ltd	16.112	1.705	Ordinary
	069	2	(in light)
an ines Australia ciu	000	7/	Orginary
UBS Securities Australia Ltd	37,980	4,019	Ordinary

UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd		268 16,023 16,023 3,407 3,407 3,407 3,407 3,078 3,078 3,078 3,078 3,078 3,078 3,078 3,078 3,078 3,078 3,078 3,078 3,078 3,078 4,099 5,099	(25) (1676 (1676) (1676	Ordinary Ordinary
UBS Securities Aust UBS Securities Aust	reale Ltd reale	See See	880 16,023 3,427 9,407 9,407 9,407 1,463 1,63 1,63 1,29 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,280	(984) (987) (1 (20) (1 (20) (1 (20) (1 (20) (1 (20) (1 (20) (2	Ordinary Ordinary
UBS Securities Aust UBS Securities Aust	ratia Itd ratia Itd	Section Sect	16,023 9,407 3,422 9,866 9,868 2,868 2,868 1,539 6,023 6,023 6,023 6,023 6,023 1,539 1,109 2,201 1,539 1,539 1,539 1,539 1,539 1,109 2,201 1,00 2,201 1,00 2,201 1,00 2,201 1,00 2,201 1,00 2,201 1,00 2,201 1,00 2,201 1,00 2,00	(1679) (1679) (1757) (1	Ordinary Ordinary
UBS Securities Austi UBS Securities Austi	rate ind rat	Sel	9,407 9,407 9,866 9,866 9,307 1,463 2,868 1,539 6,023 6,030 6,030 6,030 6,030 1,109 1,109 1,109 1,109 1,109 1,2,600 1,2	(1038) (1038) (1031) (1031) (1031) (1031) (110) (121) (121) (121) (1310)	Ordinary Ordinary
UBS Securities Austi UBS Securities Austi	ratio Ltd ratio		3,422 9866 9866 9866 1,463 3,078 1,53 1,53 1,53 1,53 1,53 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,109 1,200 1,	(359) (1530) (1530) (1530) (170) (170) (170) (170) (180) (19	Ordinary Ordinary
UBS Securities Aussi UBS Securities Aussi	ratio Ltd ratio Ltd		9.866 9.37 1,663 2,868 2,868 1,539 6,023 6,023 6,023 6,023 6,023 1,539 6,023 1,539 6,023 1,539 6,023 1,539 6,023 6,023 6,023 1,539 6,023 7,59 1,109 1,109 1,109 1,109 1,260 1,26	(1,032) (153) (153) (173) (173) (161) (161) (161) (161) (161) (161) (163	Ordinary Ordinary
UBS Securites Aust UBS Securites Aust	ratio tid ratio		937 1,463 2,868 2,903 1,539 6,033 6,030 6,080 6,080 6,080 1,109 1,109 1,109 1,109 1,2600 2,29 1,109 1,2600 1	(153) (100)	Ordinary Ordinary
UBS Securities Aust UBS Securities Aust	ratio tid ratio	Section Sect	1,463 2,868 3,078 163 2,20 2,49 2,49 2,69 2,69 2,69 2,68 2,68 2,68 2,68 2,68 2,68 2,68 2,68	(153) (100)	Ordinary Ordinary
UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi	ration to contain the containt the containt	Sel Sel	2,868 3,078 163 5,220 2,49 2,49 1,539 6,023 6,023 6,030 1,109 1,109 1,109 1,109 1,109 1,2600 2,29 1,2600 1,2	(132) (17) (17) (1846) (1617)	Ordinary Ordinary
UBS Securities Austi UBS Securities Austi	ration to consist to c		3,078 163 249 249 249 249 269 269 269 269 269 279 1,109 1,109 1,109 279 1,109 279 1,109 279 1,109 279 1,109 279 279 279 279 279 279 279 27	(322) (1322) (1323)	Ordinary Ordinary
UBS Securities Austr UBS Securities Austr	ration to ration		3,008 163 2,220 2,220 2,23 1,539 6,023 6,023 6,023 6,023 1,109 1,109 1,109 1,109 1,109 1,109 1,200	(157) (167) (167) (167) (167) (167) (167) (17) (17) (17) (17) (18) (19) (19) (19) (19) (19) (19) (19) (19	Ordinary Ordinary
UBS Securities Austrilles Austril	ratio tud ratio		5.220 2.49 2.49 2.49 1.539 6.023 8.6 1.03 2.03 1.109 1.109 1.109 1.2600	(17) (246) (400) (161) (630) (636) (636) (636) (636) (14) (151) (1	Ordinary Ordinary
UBS Securities Austructives Austructives Austructies A	rate lid rate lid		2,49 2,49 3,824 6,033 6,033 6,030 1,109 1,10 1,10	(546) (546) (400) (400) (630) (630) (638) (638) (638) (638) (116) (116) (1318)	Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust	ration to contain the contain the contain the contain the contained the		249 1,539 6,023 6,023 6,023 2,231 2,231 2,29 1,109 2,01 1,109 2,109 2,109 2,109 2,109 2,109 2,26 3,509 1,500 1,2,600 2,109 2,200 1,2,600 2,200 1,2,600 2,200 1,2,600 2,200 1,2,600 2,200 2,200 1,2,600 2,200	(161) (161) (161) (161) (163) (163) (17) (17) (17) (17) (17) (17) (17) (17	Ordinary Ordinary
UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi	raile trd raile trd		3824 (1,539 (6,02) 86 (9,02) 2,630 (1,109 (1	(400) (630) (630) (630) (636) (636) (636) (636) (116) (120) (1210) (1310	Ordinary Ordinary
UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi UBS Securities Austi	ratio Ltd ratio Ltd	S S S S S S S S S S S S S S S S S S S	6023 86 86 86 86 1,109 1	(161) (630) (638) (638) (638) (638) (638) (639) (738) (731) (7318	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust	ratio Ltd ratio Ltd	10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	6,023 86 3,231 6,080 2,680 2,78 3,785 1,109 1,109 1,2,600 1,2,600 1,2,600 1,2,600 1,2,600 1,2,600 1,2,600 1,2,600 1,3,600 1,500 1,500 1,500 1,500 1,500 1,500 1,500 1,500 1,500 1,500	(630) (338) (338) (338) (330) (14) (17) (17) (17) (17) (17) (17) (17) (18) (19) (100	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust	raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid raia Lid	Section Sect	3,231 6,080 2,080 2,29 1,109 1,109 1,500 1,200 1,200 1,500 1	(338) (338) (338) (338) (338) (438) (419)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust	raine tru raine tru	Selection (1997)	3,231 6,080 2,68 2,29 1,109 1,109 2,20 1,109 2,20 1,2,600 1,2,	(338) (638) (638) (638) (330) (10) (1318) (1,3	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust	ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud ratale tud		2.680 2.680 2.680 2.75 2.75 2.75 2.70 1.109 5.70 5.70 9.50 9.50 9.50 9.50 9.50 9.50 9.50 9.5	(636) (636) (236) (130) (110) (1318) (1318) (140) (100	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust	ratia Ind ratia Ind ratia Ind ratia Ind ratia Ind ratia Ind ratia Ind ratia Ind ratia Ind ratia Ind	S S S S S S S S S S S S S S S S S S S	268 268 3,155 229 1,109 201 1,2600 1,260 1	(28) (28) (330) (330) (310) (10) (10) (10) (10) (10) (10) (10) (Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust	rate ind rate ind rate ind rate ind rate ind rate ind rate ind rate ind rate ind rate ind rate ind rate ind rate ind rate ind	Se	268 3,158 229 1,109 1,109 201 12,600 526 526 3,509 15 15 15 15 28,173 64,960	(23) (330) (24) (116) (116) (12) (12) (12) (100)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust UBS Securities Aust	ratie tid ratie tid ratie tid ratie tid ratie tid ratie tid ratie tid ratie tid	Section Sect	3,155 229 1,109 1,260 1,260 1,260 1,360 1,36 1,509 1,5	(330) (24) (116) (1,318) (1,31	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust UBS Securities Aust	raie tid raie tid raie tid raie tid raie tid raie tid raie tid raie tid	19 20 20 20 20 20 20 20 20 20 20 20 20 20	229 1,109 201 12,600 12,600 13,4 956 3,509 115 28,173 64,960	(24) (116) (11318) (1318) (55) (100)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust UBS Securities Aust	rale ted rale ted rale ted rale ted rale ted rale ted rale ted rale ted	Self in the self i	1,109 201 12,500 526 526 956 956 3,509 1,509 28,173 28,173	(116) (21) (1,318) (55) (14) (100) (367) (12) (2,947) (6,795)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Aust	raila Itd raila Itd raila Itd raila Itd raila Itd raila Itd	Self Self Self Self Self Self Self Self	201 12,600 52,6 13,4 13,4 15,509 1,5	(1.218) (1.218) (1.218) (100) (100) (100) (12) (2.947) (6.795)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
100 100 100 100	ralia Ltd ralia Ltd ralia Ltd ralia Ltd ralia Ltd ralia Ltd ralia Ltd ralia Ltd	Sel	12,600 526 134 134 956 3,509 115 28,173	(1,318) (55) (14) (100) (100) (367) (2,547) (5,795)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1 IRS Societies Aust	raile Ltd raile Ltd raile Ltd raile Ltd raile Ltd raile Ltd raile Ltd raile Ltd	Sel 18	134 134 134 135 1350 1350 15 115 115 115 128,173	(1,516) (14) (100) (367) (2,947) (6,795)	Ordinary Ordinary Ordinary Ordinary Ordinary
PS Securities Apprehiment	rala Ltd rala Ltd rala Ltd rala Ltd rala Ltd rala Ltd	1	3,509 956 3,509 115 28,173 64,960	(35) (100) (100) (367) (2,947) (6,795)	Ordinary Ordinary Ordinary Ordinary Ordinary
Top Committee Aug	traila Ltd raila Ltd raila Ltd raila Ltd	Sel	134 956 3,509 115 28,173 64,960	(14) (100) (367) (12) (2,947) (6,795)	Ordinary Ordinary Ordinary Ordinary
Obs securities Australia Ltd	raia Ltd raia Ltd raia Ltd	Sell Sell Sell Sell Sell Sell	956 3,509 115 28,173 64,960	(100) (367) (12) (2,947) (6,795)	Ordinary Ordinary Ordinary
UBS Securities Australia Ltd	rais Ltd rais Ltd rais Ltd	3. 3. 3. 3. 3. 3. 3.	3,509 115 28,173 64,960	(367) (12) (2,947) (6,795)	Ordinary Ordinary Ordinary
UBS Securities Australia Lto	ralia Ltd ralia Ltd ralia Ltd		28,173 64,960	(12) (2,947) (6,795)	Ordinary Ordinary
UBS Securities Aust	ralia Ltd ralia Ltd	Sell Sell Sell Sell	28,173	(2,947)	Ordinary
UBS Securities Aust	ralia Ltd	Sell Sell Sell	64,960	(6,795)	
UBS Securities Aust		Sell			Ordinary
UBS Securities Australia Ltd	rala Ltd	Sell	5 640	(065)	Ordinary
UBS Securities Aust	ralia Ltd		9.856	(1 031)	Ordina
UBS Securities Australia I to	ralia I td	-	17 122	(107.1)	Coding
1 IRS Societies Aust	ralia I+d	000	4.04.	(1,7,7)	1000
COS SCOURS ASSUBILIES COS	and Ltd	190	4.0.4	(Doc,1)	Orginary
ODS Securities Aust	Jana Liu	Sell.	34,292	(3,587)	Ordinary
Upo securities Australia Ltd	ralla Ltd	Sell	4,780	(200)	Ordinary
UBS Securities Australia Ltd	ralia Ltd	Sell	4,952	(518)	Ordinary
UBS Securities Aust	ralia Ltd	Sell	1,922	(201)	Ordinary
UBS Securities Aust	ralia Ltd	Setl	153	(16)	Ordinary
UBS Securities Aust	ralia ttd	Seil	1.797	(188)	Ordinary
UBS Securities Australia Ltd	ralia Ltd	Sell	15.258	(1 596)	Ordinary
UBS Securities Austr	ralia I td	llaS	92	(8)	Carion
I IRS Socientias Aust	rain Ital	Coll	0.00	9 5	o o o
I IRS Securities Australia I to	rain I+d	100	17 905	(C)	Ordinary
I IRS Securities Arist	ralia Ltd	illo Call	060'/1	(1,0/2)	Collingia
UBS Securities Australia Ltd	ralia I td		1 204	(301)	Ordinary
I IRS Securities Australia 14d	ralia I tol	130	100,07	(152.7)	Ordingry
I IRS Securities Austr	in the	li S	120,27	(1,00,1)	Ordinary
BS Securities Australia 114	ralia ttd	ll C	14.246	(3,714)	Ordinary
I IBC Cocurition Aurit	Par pilo	100	14,040	(1)300()	Ordinary
1 IDC Con mitter August	alia cita	Jac 1	+7c/cc	(5)/14/	Ordinary
OBS SECULIES AUSTRALIA LIG	lana Ltd	Table 1	719	(64)	Ordinary
I DC C-	מוש רוס	lak l	36,165	(3,/81)	Ordinary
UBS Securities Australia Ltd	rana Ltd	Sell	35,524	(3,714)	Ordinary
UBS Securities Australia Ltd	ralia Ltd	Sell	2,868	(300)	Ordinary
UBS Securities Australia Ltd	ralia Ltd	Sell	5,832	(010)	Ordinary
UBS Securities Austr	Australia Ltd	Sel}	870	(16)	Ordinary
UBS Securities Australia Ltd	ralia Ltd	Sell	1,367	(143)	Ordinary
UBS Securities Australia Ltd	ralia Ltd	Sell	1,807	(189)	Ordinary
UBS Securities Australia Ltd	ralia Ltd	Sell	727	(92)	Ordinary
UBS Securities Australia Ltd	ralia Ltd	Sell	535	(20)	Ordinary
UBS Securities Australia Ltd	alia Itd	las	201	(14)	Ordinan
I IBS Sociarities Austr	41	170	7 440	24.57	(in line)
UED SECULINES MUSIC		lies	611,	(744)	Ordinary

Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary
(177)	(146)	(7,515)	(256)	(816,1)	(4,209)	(1,280)	(5,130)	(422)	(515)	(499)	(135)	(011)	(1.159)	(809)	(400)	(1,786)	(1,147)	(3,447)	(65)	(687)	(400)	(500)	(2,707)	(2,500)	(2,518)	(45)	(5,531)	(192)	(1,461)	(20)	(110)	(330)	(142)	(120)	(1,500)	(271)	(3,887)	(118)	6	(330)	(9)	(330)	(111)	(33)	(54)	(22)	(342)	(7,553)	(206)	(330)	(8)	(946)	(376)	(797)	(252)	(1/8)	(160)		(234)
1,692	1,396	71,843	5,029	18,336	40,238	12,237	49,043	4.034	4,923	4,770	1 291	1 052	11.080	5.812	3.824	17,074	10,965	32,953	621	6,568	3,824	4.780	25,879	23,900	24,072	430	52,8/6	1.836	13,967	191	1,052	3,155	1,358	1,148	14,348	2,591	37,160	1,128	29	3,155	57	3,155	1,061	315	516	210	3,270	72,207	4,837	3,155	92	9,044	3,595	7,619	2,409	1,702	8 547	8,547	1777
=	=	=		_	=	=											11							_																						-													_
leS.	leS.	<u>₩</u>	Sel	Sel	Sel	Sel	Sel	JaS.	Sel	es	3.5	le2	3	Sell			Sel		Sell		la Co				lles		× 3					<u> </u>			Seil	Sel	les Sell	Sell			Sell		Jel Sel		Sell			Sell	Sell		¥ 3	Sells	Sel	Sel	8 3	S S	oc oc	Sec.	2
S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	5 Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Etd	S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Contribute Attended 110
11-Jul-14 UB:	T			_		11-Juf-14 UB:	Γ	Γ	Γ	T	T	T	Ť		Į.				Ī		11-Jul-14 UB:		1		- 1	-1	ŀ	11-Jul-14 UB	1	П	- 1	11-Jul-14 UBS	1		П	T	11-Jul-14 UBS	Т	11-Jul-14 UBS	П	11-Jul-14 UBS	-	11-5ul-14 UBS		l	Н		11-Jul-14 UBS	T	11 - 11 14 UBS	Ť	Ť	Ħ		T	T	11-1ui-14 UBS	T	

100					200
11.51.11	I IBC Securities Australia tid	lec Sell	4,114	(433)	Ordinary
1-101-14	URS Securities Australia Ltd	Bell	3,781	(398)	Ordinary
1-111-14	IRS Societies Australia 144	in d	9,400	200,	Ordinary
11.1.1	(DC Countries Assertial Ltd	Sec.	505'/	SOS	Ordinary
1 1 1	Upo Securities Australia Ltd	Boy	301	32	Ordinary
+1-101-1	Obs securities Australia Ltd	Buy	301	32	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	8,422	896	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	1,617	172	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	2,312	246	Ordinary
1-Jul-14	UBS Securities Australia Ltd	8ny	4,127	439	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buv	7.229	769	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Bliv	900 9	660	Ording
1-11-14	LIRS Securities Australia 11d	in a	1081	3	Countary
1.1.1.14	I IBS Sea prition Americal and	660	100,1	0 1	Cramary
100	ODS Securities Australia Ltd	any	7,397	555	Ordinary
-101-14	UBS Securities Australia Ltd	Buy	4,315	459	Ordinary
1-101-14	UBS Securities Australia Ltd	Buy	8,714	927	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	2,670	284	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buy	17,221	1,832	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Buv	8.874	770	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Sirv	5 151	248	Ordinary
1-Jul-14	UBS Securities Australia Ltd	Biv	113	12	Ordinan
1-14-14	I IBS Securities Australia I td	, and	0000	7,000	Cidiliary
1.1.1.1	I IDC Committee Assertial to	ong.	2,049	210	Ordinary
100	Obs securities Australia Ltd	Buy	7,858	836	Ordinary
1-JUI-14	+	Buy	10,161	1,081	Ordinary
3/06/2014	_	Sell	42,179	(4,304)	Ordinary
3/06/2014	-1	Sell	490,000	(20,000)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Buy	1.392	142	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sel	12.358	(1.261)	Ordinary
3/06/2014	I IBS Sequirities Australia I to	= 0	000.01	(107)	9 9
2/06/2014	1	i c	000,000	(5,102)	Ordinary
2,000,2014		Sell	/90'5	(/1<)	Ordinary
3/U5/2014	UBS Securities Australia Ltd	Seil	1,186	(121)	Ordinary
3/06/2014	_	Sell	83,261	(8,496)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	44.129	(4, 503)	Ordinary
3/06/2014	╀	Coll	010 100	(200,00)	
1000017	+	100	656,432	(26,935)	Ordinary
200000	+	Jac.	8,820	(Ann)	Ordinary
5/05/2014	+	- Ke	000'86	(10,000)	Ordinary
3/06/2014	_	Sell	241,580	(24,651)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	14.641	(1 494)	Ordinary
3/06/2014	┝	llas	135,633	(12.926)	Judion
3/06/2014		Coll	200	1007	Contrary of
1000000		1100	45.7	(ne)	Ordinary
3/U6/2014	UBS Securities Australia Lto	Self	88	(6)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	294	(30)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	81,757	(8.334)	Ordinary
3/06/2014	UBS Securities Australia Ltd	llay	4 905	(500)	Ordings
3/06/2014	I IBC Counting Australia 144	100	206,4	(000)	Coming
100000	Dispersion of the second of th	P. C.	10,634	(1,084)	Ordinary
3/05/2014	UBS Securities Australia Ltd	Set	3,924	(400)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	4,316	(440)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	4,777	(487)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	3,071	(313)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	6 749	(688)	Ordinary
3/06/2014	URS Securities Australia 11d	9	40	(2)	(in the last
3/06/2014	I IBC Contribut Australia 14d	100	64	(6)	orange F
10000000	The state of the s	IIAC II	0/0'61	(386,1)	Ordinary
20072014	4	Seil	3,9/3	(405)	Ordinary
3/06/2014	7	Sell	3,492	(326)	Ordinary
3/06/2014	~	Sell	893	(16)	Ordinary
3/06/2014	_	Sell	1,923	(196)	Ordinary
3/06/2014	UBS Securities Australia Ltd	ile?	373	(38)	Ordinan
3/06/2014	UBS Securities Australia 1 td	9	2008	(10)	Ordinary
13/06/2014	I IDC Complian Australia (+)	===	05000	(16)	Courain
400000	Obs securities Australia Ltd	Jac Jac Jac Jac Jac Jac Jac Jac Jac Jac	16,3/3	(1,669)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	294	(30)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	6,524	(999)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Sell	4.817	(491)	Ordinary
1/06/2014	UBS Securities Australia Ltd	les	61891	(605.9)	Ordinary
3/06/2014	UBS Securities Australia Ltd	Ses	2.433	(202)	Ordinary
13/06/2014		-	17114		
1	IRS Securities Australia ted	100	190,00	10,0	O'dillially

11-bb-14 UBS Securities Australia Ltd 1-bb-14 UBS Securities Australia	Australie Ltd Australie Ltd		3,117 1,836 1,13,149 201 201 201 76,977 76,9	(326) (102,914) (102,914) (102,914) (107,748)	Ordinary Ordinary
	Australia Ltd Australia Ltd		1,836 123,488 123,488 3,164 3,165 3,155 1,155 1,166 1,	(129) (12,914) (12,914) (10,748) (10,74	Ordinary Ordinary
	Australia Ltd Australia Ltd		1,3458 1,1458 1,1458 1,155 1,155 1,155 1,156 1,165 1,165 1,165 1,165 1,185	(123) (107,48) (107,4	Ordinary Ordinary
	Australia Ltd Australia Ltd		123,488 164 3 164 201 76,977 76,977 76,977 77,763 201 202 220 2245 287 287 287 287 287 287 287 287 287 287	(12,914) (12,914) (12,914) (13,014) (13	Ordinary Ordinary
	Australia Ltd Australia Ltd		3,164 76,977 76,977 7,763 3,155 102,751 306,75 2,495 2	(320) (10,748) (320) (10,748) (321)	Ordinary Ordinary
	Australia Ltd Australia Ltd		201 7,637 7,763 3,155 10,751 206 20 20 2,645 2,679 2,679 2,679 1,166 1,166 1,165 2,046 1,185 1,185 1,185 1,185 1,185 1,185 3,126 3,126 3,126 1,185 1,1	(123) (123)	Ordinary Ordinary
	Australie Itd Australie Itd		76,977 76,977 77,783 7,783 7,783 7,082 7,095 7,095 7,095 7,095 7,095 7,095 7,096 7,0	(8,052) (10,748) (10,	Ordinary Ordinary
	Australia Ltd Australia Ltd		7,763 3,155 102,751 2,495 2,495 2,495 2,495 2,495 2,495 2,495 2,495 2,495 2,495 2,495 2,495 3,155 1,166	(10,748) (10	Ordinary Ordinary
	Australia Ltd Australia Ltd		3,155 102,751 306 306 220 2,245 2,679 3,155 1,166 1,166 1,166 1,635 3,155 1,635 3,155 1,635 3,155 1,635 1,635 3,155 1,635	(152) (152) (153) (154)	Ordinary Ordinary
	Australia Ltd Australia Ltd		102/51 206 220 220 2,495 3,155 1,1696	(10,748) (23) (23) (261) (261) (263) (1,669) (1,669) (1,601) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003)	Ordinary Ordinary
	Australia Ltd Australia Ltd		306 2495 2495 287 287 287 287 3155 1166 168 1685 1685 1085 1085 1085 1085 1185 1185 1185 1185 1185	(32) (30) (30) (30) (152) (152) (152) (152) (152) (152) (153) (153) (153) (153) (153) (153) (153) (153) (153) (153)	Ordinary Ordinary
	Australia Ltd Australia Ltd		220 287 287 287 287 3,679 1,166 1,166 1,635 3,824 3,824 3,125 2,046 1,485 1,185 3,126 3,12	(51) (150) (Ordinary Ordinary
	Australia Ltd Australia Ltd	8	2,495 2,679 2,679 3,155 11,595 1,166 1,166 1,616 1,616 1,635 3,824 402 230 402 210 210 210 210 210 210 210 2	(150) (150) (150) (150) (150) (150) (150) (150) (150) (150) (150) (150) (150) (150) (150) (150) (150) (150)	Ordinary Ordinary
	Australia Ltd Australia Ltd		287 3,155 3,155 1,166 1,166 1,166 1,635 1,63	(122) (1,003) (123) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003)	Ordinary Ordinary
	Australia Ltd Australia Ltd		5,679 15,956 1,166 1,166 1,635 1,635 1,635 1,635 1,035 1,156 1,185 1,185 3,126	(52) (1,691) (1,691) (1,691) (1,691) (1,691) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003)	Ordinary Ordinary
	Australia Ltd Australia Ltd		3,155 15,956 1,166 1,166 1,635 3,824 402 210 210 2,046 1,185 1,185 3,126 3,126 3,126	(129) (120) (120) (120) (120) (100) (100) (100) (100) (100) (100) (100) (100) (100) (100) (100)	Ordinary Ordinary
	Australia Itd Australia Itd		15,956 1,166 1,666 16,168 1,635 1,635 1,035 1,135 1,185 1,185 1,185 1,185 1,185	(1,669) (1,669) (1,691) (1,601) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003)	Ordinary Ordinary
	Australia Ltd Australia Ltd	8 18 18 18 18 18 18 18 18 18 18 18 18 18	1,166 239 1,635 1,635 1,635 1,035 2,046 1,453 1,185 3,126 3,	(1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003) (1,003)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia tud Australia tud		1,000 1,010 1,035 1,035 1,035 1,036 1,145 1,145 1,185 1,185	(122) (1,691) (1,691) (171) (171) (1,691) (1,691) (1,691) (1,691) (1,692) (1,693) (1,693) (1,693) (1,693) (1,693) (1,693) (1,693) (1,693)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia ttd Australia ttd		16,166 1,635 1,824 402 210 2,89 3,155 1,145 1,185 3,126 3,126 3,126 3,126 3,126 3,126 3,126	(1, 691) (171) (171) (171) (400) (42) (42) (120) (134) (152) (152) (152)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia ttd Australia ttd	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	1,635 3,824 402 200 2,89 9,589 9,155 2,046 1,453 1,185 3,126 3,126	(400) (400) (400) (400) (100) (100) (100) (120) (120)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia Ltd Australia Ltd	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	3,824 402 402 2,10 9,589 3,155 1,185 3,126 3,126 3,126	(124) (124) (1,003) (130) (152) (124) (124)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia Itd Australia Itd	Sell Sell	402 402 210 9,589 3,155 2,046 1,453 1,185 3,126	(42) (22) (1,003) (1,003) (214) (214) (152) (152) (152)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia tid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid Australia Lid	Test See	210 9,589 3,155 2,046 1,453 1,453 3,126	(1,003) (1,003) (330) (214) (152) (124)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	a S	9,589 3,155 2,046 2,185 1,185 3,126	(1,02) (330) (330) (214) (152) (124)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	100	3,155 2,046 1,453 1,185 3,126	(330) (214) (152) (124)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd		2,046 1,453 1,185 3,126	(152) (152) (124)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	Joseph	1,453 1,185 3,126	(152)	Ordinary Ordinary Ordinary Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	Sall	1,185	(124)	Ordinary Ordinary Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd		3,126	(124)	Ordinary Ordinary Ordinary
1 1 1 1 1 1	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	3	2,120		Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	100	300 38	(175)	Ordinary
	Australia Ltd Australia Ltd Australia Ltd Australia Itd	1100	45,053	(4,7,17)	Ordinary
	Australia Ltd Australia Ltd Australia Ltd	- N	25,000	(001)	
	Australia Ltd Australia Ltd Anetralia Itd	les l	117,12	(2,847)	Ordinary
	Australia Ltd Anetralia I td	Je J	3,422	(358)	Ordinary
	Ailetralia I 1d	Jell Sell	3,403	(326)	Ordinary
Π	Oddona rice	Sell	3,155	(330)	Ordinary
П	Australia Ltd	Sell	48	(5)	Ordinary
П	Australia Ltd	Sell	9/	(8)	Ordinary
Γ	Australia Ltd	Sell	363	(38)	Ordinary
_	UBS Securities Australia Ltd	Sell	315	(33)	Ordinary
1-Jul-14 UBS Securities Australia Ltd	Australia Ltd	les	53	(3)	Ordinary
	Australia Ltd	Sell	249	(26)	Ordinary
Ī	Australia Ltd	195	73.301	(2 445)	Ordinary
ı	Australia Ltd	lles	23 310	(2) 446)	Ordinan
l	Australia Ltd	Sell	3 154	(331)	Ordinary
	Australia I to	100	4,000	(100)	Coming
1	Australia Ltd	100	4,036	(430)	Ordinary
T	Australia Liu	ne o	2	(I)	Ordinary
Ť	Australia Ltd	Sell	3,154	(331)	Ordinary
1-Jui-14 UBS Securities Australia Ltd	Australia Ltd	Ze#	6,061	(989)	Ordinary
1-Jul-14 UBS Securities /	Australia Ltd	Seli	15,972	(1,676)	Ordinary
ul-14 UBS Securities Australia Ltd	Australia Ltd	Sell	2,049	(515)	Ordinary
1-Jul-14 UBS Securities Australia Ltd	Australia Ltd	Sell	3,812	(400)	Ordinary
11-Jul-14 UBS Securities Australia Ltd	Australia Ltd	Sell	3.154	(331)	Ordinary
Ī	Australia Ltd	Sell	248	(36)	Ordinary
Γ	Australia Ltd	ق	17 691	(3 055)	Ordinary
1-Jul-14 LIBS Securities Australia Ltd	Australia I td	و	4 008	(CCC,C)	Ordinary
Ť	A. citation A.	1170	4,000	(420)	Clumary
Ť	Australia Ltd	liac J	060,4	(430)	Orginary
T	Australia Ltd	36	67	(F)	Ordinary
9	Australia Ltd	E S	67	(3)	Ordinary
BS	Securities Australia Ltd	3	884	(63)	Ordinary
	Australia Ltd	Seti	53	(3)	Ordinary
	Australia Ltd	Seli	096	(101)	Ordinary
UBS Securit	Australia Ltd	Sell	52	(8)	Ordinary
	ies Australia Ltd	Sell	836	(88)	Ordinary
Γ	Australia I td	ll d'S	200	(21)	Ordinan
11-Jul-14 Obs Securities Australia Liu	אמאון שווש דונט	liac liac	707	(17)	Ordinary

13/06/2014	UBS Securities Australia Ltd	Sell	66,316	(6,760)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	244,227	(24,883)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	2/29	(69)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	8,790	(968)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	49	(5)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	31,784	(3,240)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	34,786	(3,546)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Self	5,739	(282)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Seli	8,790	(968)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	3,924	(400)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	8,132	(628)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	78	(8)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	3,924	(400)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	299	(89)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	491	(20)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	12,753	(1,300)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	481	(64)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	20,984	(2,139)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	25,016	(2,550)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	893	(16)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	47,225	(4,814)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	6,887	(202)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	4,964	(905)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	38	(4)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	481	(49)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	20,012	(2,040)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	3,159	(322)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	1,148	(117)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	695	(85)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	294	(30)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	363	(37)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	11,017	(1,123)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	10	(1)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Sell	26,813	(2,736)	Ordinary
13/06/2014	UBS Securities Australia Ltd	Buy	26,813	2,736	Ordinary